Desktop Computer Displays:

A Life-Cycle Assessment



VOLUME 1

Maria Leet Socolof Jonathan G. Overly Lori E. Kincaid Jack R. Geibig



This document was produced by the University of Tennessee Center for Clean Products and Clean Technologies under grant #82537401 from EPA's Design for the Environment Branch, Economics, Exposure, & Technology Division, Office of Pollution Prevention and Toxics.

EXECUTIVE SUMMARY

This report presents the results of a voluntary, cooperative project among the Design for the Environment (DfE) Program in the Economics, Exposure, and Technology Division of the U.S. Environmental Protection Agency's (EPA) Office of Pollution Prevention and Toxics, the University of Tennessee (UT) Center for Clean Products and Clean Technologies, the electronics industry, and other interested parties to develop a model and assess the life-cycle environmental impacts of flat panel display (FPD) and cathode ray tube (CRT) technologies that can be used for desktop computer displays. The DfE Computer Display Project (CDP) report provides a baseline analysis and the opportunity to use the model as a stepping stone for further analyses and improvement assessments for these technologies.

The DfE CDP uses life-cycle assessment (LCA) as an environmental evaluation tool that looks at the full life cycle of the product from materials acquisition to manufacturing, use, and final disposition. As defined by the Society of Environmental Toxicology and Chemistry, there are four major components of an LCA study: goal definition and scoping, life-cycle inventory, impact assessment, and improvement assessment. The more recent International Standards Organizations definition of LCA includes the same first three components, but replaces the improvement assessment component of LCA with a life-cycle interpretation component. LCAs are generally global and non-site specific in scope.

The DfE CDP analysis also incorporates some elements of the Cleaner Technologies Substitutes Assessment (CTSA) methodology (Kincaid *et al.*, 1996), which was developed under the DfE Program to help businesses make environmentally informed choices and design for the environment. The CTSA process involves comparative evaluations of the relative human and ecological risk, energy and natural resource use, performance, and cost of substitute technologies, processes, products, or materials.

This project focuses on the LCA, while including some CTSA-related analyses. It performs the broad analysis of the LCA, which also incorporates many of the CTSA components (e.g., risk, energy impacts, natural resource use) into the impact assessment. The analysis also assesses more specific impacts for selected materials and acknowledges product cost and performance, typical of a CTSA. As only selected materials are qualitatively evaluated for the CTSA, this project is an LCA with a streamlined CTSA component.

LCAs evaluate the environmental impacts from each of the following major life-cycle stages: raw materials extraction/acquisition; materials processing; product manufacture; product use, maintenance, and repair; and final disposition/end-of-life. The inputs (e.g., resources and energy) and outputs (e.g., products, emissions, and waste) within each life-cycle stage, as well as the interaction between each stage (e.g., transportation) are evaluated to determine the environmental impacts.

In this study and project report, the goal and scope of the CDP are the subject of Chapter 1. The life-cycle inventory (LCI), which involves the quantification of raw material and fuel inputs, and solid, liquid, and gaseous emissions and effluents, is the subject of Chapter 2. The life-cycle impact assessment (LCIA) involves the translation of the environmental burdens identified in the LCI into environmental impacts and is the subject of Chapter 3. The improvement assessment or life-cycle interpretation is left to the electronics industry given the results of this study. The report also includes a qualitative risk screening of selected materials to represent the CTSA component of the report in Chapter 4. The summary and conclusions are presented in Chapter 5.

I. GOAL DEFINITION AND SCOPE

Purpose and Need

The purpose of this study is two-fold: (1) to establish a scientific baseline that evaluates the life-cycle environmental impacts of active matrix liquid crystal display (LCD) and cathode ray tube (CRT) technologies for desktop computers, by combining LCA and CTSA methodologies; and (2) to develop a model that can be used with updated data for future life-cycle analyses. This study is designed to provide the electronics industry with information needed to improve the environmental attributes of desktop computer displays. The evaluation considers impacts related to material consumption, energy, air resources, water resources, landfills, human toxicity, and ecological toxicity. It is intended to provide the industry and consumers with valuable information to make environmentally informed decisions regarding display technologies, and enable them to consider the relative environmental merits of a technology along with its performance and cost. While there has been some work done on the life-cycle environmental impacts of either CRTs or LCDs, there has not been a quantitative LCA of both CRTs and LCDs.

At present, computer displays using CRTs dominate worldwide markets. The LCD, first used predominately in notebook computers, is now moving into the desktop computer market. CRTs use larger amounts of energy to operate than LCDs, and are associated with disposal concerns due to leaded glass in the displays. LCDs may consume more energy during manufacturing and contain small amounts of mercury. Given the expected market growth of LCDs for computer displays, the various environmental concerns throughout the life cycle of the computer displays, and the fact that the relative life-cycle environmental impacts of LCDs and CRTs have not been scientifically established to date, there is a need for an environmental lifecycle assessment of both of these types of desktop computer display technologies.

Targeted Audience and Use of the Study

The electronics industry is expected to be one of the primary users of the study results. The study is intended to provide industry with an analysis that evaluates the life-cycle environmental impacts of selected computer display technologies. Another result of the study is an accounting of the relative environmental impacts of various components of the computer displays, thus identifying opportunities for product improvements to reduce potential adverse environmental impacts and costs. Since this study incorporates a more detailed health effects component than in traditional LCAs, the electronics industry can use the tools and data to evaluate the health, environmental, and energy implications of the technologies. With this evaluation, the U.S. electronics industry may be more prepared to meet the demands of extended product responsibility that are growing in popularity in the global marketplace, and better able to meet competitive challenges in the world market. In addition, the results and model in this study will provide a baseline LCA upon which alternative technologies can be evaluated. This will allow for more expedited display-related LCA studies, which are growing in popularity by industry and may be demanded by original equipment manufacturers (OEMs) or international organizations.

EPA and interested members of the public can also benefit from the results of the project. The project has provided a forum for industry and public stakeholders to work cooperatively, and the results can be used by stakeholders as a scientific reference for the evaluated display technologies. The results of the project could also be of value to other industries involved in designing environmental improvements into the life cycle of consumer products.

Product System

The product system being analyzed in this study is a standard desktop computer display that functions as a graphical interface between computer processing units and users. Besides the CRT display, several FPD technologies were considered for inclusion in this study. Among the FPD technologies that exist, the amorphous silicon (a:Si) thin-film transistor- (TFT) active matrix LCD technology meets the requirements of the functional unit within the parameters of this analysis and is assessed in this study.

The product system is the computer display itself and does not include the central processing unit (CPU) of the computer that sends signals to operate the display. It is assumed that the LCDs operate with an analog interface, and therefore are compatible with current CRT CPUs as plug-and-play alternatives.

In an LCA, product systems are evaluated on a functionally equivalent basis. The functional unit is used as the basis for the inventory and impact assessment to provide a reference to which the inputs and outputs are related. For this project, the functional unit is one desktop computer display over its lifespan, which meets the functional unit specifications presented in Table ES-1. The CRT technology is the current industry standard for this product system.

Specification	Measure
display size ^a	17" (CRT); 15" (LCD)
diagonal viewing area ^a	15.9" (CRT); 15" (LCD)
viewing area dimensions	12.8" x 9.5" (122 in ²) (CRT); 12" x 9" (108 in ²) (LCD)
resolution	1024 x 768 color pixels
brightness	200 cd/m ²
contrast ratio	100:1
color	262,000 colors

Table ES-1. Functional unit specifications

^a An LCD is manufactured such that its nearest equivalent to the 17" CRT display is the 15" LCD. This is because the viewing area of a 17" CRT is about 15.9 inches and the viewing area of a 15" LCD is 15 inches. LCDs are not manufactured to be exactly equivalent to the viewing area of the CRT.

Assessment Boundaries

In a comprehensive cradle-to-grave analysis, the display system includes five life-cycle stages: (1) raw materials extraction/acquisition; (2) materials processing; (3) product manufacture; (4) product use, maintenance and repair; and (5) final disposition/end-of-life. Also included are the activities that are required to affect movement between the stages (e.g., transportation).

The geographic boundaries of this assessment depend on the life-cycle stage. This LCA focuses on the U.S. display market; therefore, the geographic boundary for the use and disposition stages of displays is limited to the United States. The geographic boundaries for raw material extraction, material processing, and product manufacture are worldwide (although actual product manufacturing data were only collected from the United States, Japan, and Korea, described in Chapter 2 of the report). While the geographic boundaries show where impacts might occur for various life-cycle stages, traditional LCAs do not provide an actual spatial relationship of impacts. That is, particular impacts cannot be attributed to a specific location. Rather, impacts are generally presented on a global or regional scale.

Considering the temporal boundaries, this study addresses impacts from the life cycle of a desktop computer display manufactured using 1997-2000 technology. The use and disposition stages cover a period that represents the life of a display. The lifespan, labeled as the "effective" life, is defined as the period of time the display is in use by primary, secondary, or even tertiary users before reaching its final disposition. The effective life, used as the baseline scenario, is estimated based on past and current use patterns of displays and represents a realistic estimate of the lifespan. As the effective life is subject to many variables, including fluctuating market trends, an alternative lifespan is presented in a sensitivity analysis. The alternative lifespan, or "manufactured" life, defined as the designed durability of a display (e.g., the time a display or key display component will operate before failing), is approximated based on the manufacturer's estimated durability of the display.

Impacts from the infrastructure needed to support the manufacturing facilities (e.g., maintenance of manufacturing plants) are beyond the scope of this study. However, maintenance of clean rooms used in the manufacturing of LCDs (and other components), which require substantial amounts of energy, are considered part of the manufacturing process.

Impacts from the transportation and distribution of materials, products, and wastes throughout the life-cycle of a display were originally included in the scope of the CDP LCA. However, only a small part of the overall transport in the life of a monitor was either reported in primary data collected for this project or available in secondary data. Inconsistencies between primary and secondary transportation data sources and the overall poor quality of transport data prevented an accurate assessment of the transportation inventory and impacts. Therefore, transportation impacts were excluded from the analysis. Section 2.6 describes transport data limitations and uncertainties in detail.

II. LIFE-CYCLE INVENTORY (LCI)

General Methodology

An LCI is the identification and quantification of the material and resource inputs and emission and product outputs from the unit processes in the life cycle of a product system. For the DfE CDP, LCI inputs include materials used in the computer display product itself, ancillary materials used in processing and manufacturing the displays, and energy and other resources consumed in the manufacturing, use, or final disposition of the displays. Outputs include products, air emissions, water effluents, and releases to land. Figures ES-1 and ES-2 show the unit processes that are included in the scope of this project for the CRT and LCD life cycles, respectively.



Figure ES-1. CRT linked processes



Figure ES-2. LCD linked processes

Data were also collected on the final disposition of emissions outputs, such as whether outputs are released directly to the environment, recycled, treated, and/or disposed. This information helps determine which impacts will be calculated for a particular inventory item. Methods for calculating impacts are discussed in Chapter 3, Life-cycle Impact Assessment.

Given the enormous amount of data involved in inventorying all of the inputs and outputs for a product system, decision rules, based on the mass, environmental, energy, and functional significance, were used to determine which materials or unit processes to include in the LCI. Decision rules are designed to make data collection manageable while still representative of the product system and its impacts. Data were collected from both primary and secondary sources. Table ES-2 lists the types of data (primary or secondary) used for each life-cycle stage in the CDP LCI. In general, greater emphasis was placed on collecting data and/or developing models for the product manufacturing, use, and end-of-life life-cycle stages.

Life-cycle stage	Data types
Upstream (materials extraction and processing)	Secondary data.
Product and component manufacturing	Primary data, except secondary data used for frit.
Use	Modeled using secondary data; maintenance and repair are not included in the analysis.
Final disposition (recycling and/or disposal)	Modeled using secondary data plus primary data from CRT recycling facilities.
Packaging, transportation, distribution	Not included.

Table ES-2. Data types by life-cycle stage

In the CDP LCI, data were allocated to the functional unit (i.e., a desktop computer display over its lifetime) as appropriate. The data that were collected for this study were either obtained from questionnaires developed for this project (i.e., primary data) or from existing databases (i.e., secondary data). LCI data were imported into a Life-Cycle Design Software Tool developed by the UT Center for Clean Products and Clean Technologies with funding from the EPA Office of Research and Development and Saturn Corporation. The UT Life-Cycle Design Software Tool organizes data in such a way that each process inventory is independent. Customized "profiles" (e.g., the manufacture of a CRT or the whole life-cycle of an LCD) can be developed by linking processes.

LCI data quality was evaluated based on the following data quality indicators (DQIs): (1) the source type (i.e., primary or secondary data sources); (2) the method in which the data were obtained (i.e., measured, calculated, estimated); and (3) the time period for which the data are representative. Any proprietary information required for the assessment was aggregated to protect confidentiality.

A critical review process was maintained in the CDP LCA to help ensure that appropriate methods were employed and study goals were met. A project Core Group and Technical Work Group, both consisting of representatives from industry, academia, and government, including EPA's DfE Work Group, provided critical reviews of the assessment. The Core Group served as the project steering committee and was responsible for approving all major scoping assumptions and decisions. The Technical Work Group and EPA's DfE Work Group provided technical guidance and were given the opportunity to review all major project deliverables, including the final LCA report.

Upstream Life-cycle Stage Methodology

The materials extraction and processing inventories for key materials were obtained from a secondary LCI database developed by *Ecobilan* (1999). The U.S. electric grid inventory was developed from secondary sources by UT. The U.S. electric grid inventory was then modified, based on the distribution of fuels used in Japan, to develop the Japanese electric grid, which was used where manufacturing occurs in Asia. Electricity consumed in the life-cycles of the monitors was linked to the inventory of inputs and outputs from the U.S. or Japanese electric grid inventories, as appropriate.

Manufacturing Stage Methodology

The inventories for the product manufacturing life-cycle stage were developed from primary data collected from manufacturers in Asia and the United States. The manufacturing processes included in the study, as well as the number of data sets for each process and the country of origin of the data, are presented in Table ES-3. A total of 27 product manufacturing questionnaires were collected for 11 different processes. Allocation of data to the functional unit was conducted as necessary. Processes for which we collected more than one company's data were averaged together.

The quality of the manufacturing stage data can be evaluated against two factors: (1) the date of the data; and (2) the type of data (i.e., measured, calculated or estimated). The data collection phase of this project began in 1997 and extended through 2000. Some processes are more sensitive to production dates than others. Most processes included in this analysis are mature technologies and are not expected to differ significantly between the years 1997 and 2000. However, an exception is LCD panel/module manufacturing, which is an evolving and rapidly advancing process and has seen changes between these years. For the LCD panel and module manufacturing process, most data were from 1998 and 1999.

Data quality indicators were developed based on whether data were measured, calculated, or estimated, as reported in company data questionnaires. The weighted average of data collected and their associated DQIs are as follows: for the CRT, 43% of the data were measured, 34% calculated, 13% estimated, and 10% were not classified. For the LCD, a similar distribution shows 33% measured, 30% calculated, 23% estimated, and 14% not classified.

Product Use Life-cycle Stage Methodology

The baseline analysis in this project employs an effective life use stage scenario (the actual amount of time a monitor is used, by one or multiple users, before it is disposed of, recycled, or re-manufactured). A manufactured life scenario (the amount of time either an entire monitor or a single component will last before reaching a point where the equipment no longer functions, independent of user choice) is evaluated in a sensitivity analysis.

1	1
Process	Country of origin of data (# of data sets)
CRT monitor assembly	Japan (2), U.S. (1)
CRT (tube) manufacturing	Japan (2), U.S. (1)
CRT leaded glass manufacturing	Japan (1), U.S. (2)
CRT frit manufacturing	generic secondary data from the U.S.
LCD monitor assembly	Japan (2)
LCD panel and module manufacturing ^a	Japan (5), Korea (2)
LCD - glass manufacturing	Japan and U.S. (1) ^b
LCD - color filter patterning on front glass	Japan (1)
LCD - liquid crystal manufacturing	Japan (2)
LCD - polarizer manufacturing	Japan (1)
LCD - backlight unit assembly	Japan (3)
LCD - backlight light guide manufacturing	Japan (1)
LCD - cold cathode fluorescent lamp (CCFL) manufacturing	Japan (1)
PWB manufacturing (for CRT and LCD monitors)	generic secondary data from the U.S.

 Table ES-3. Location of companies and number of process data sets

^a The LCD *panel* consists of two glass panels patterned with transistors and color filters, liquid crystals inserted between the panels, and associated row and column drivers. The LCD *module* consists of the panel, backlight unit, and associated electronic boards for the entire panel, and the backlight.

^b The average of three data sets for CRT leaded glass manufacturing was modified to remove lead from the inventory.

To develop the use stage inventory, energy use rates [e.g., kilowatts (kW)] were combined with the time a desktop monitor is on during its lifespan (hours/life) to calculate the total quantity of electrical energy consumed during the use life-cycle stage [e.g., kilowatthours (kWh)/life]. This was then combined with the electric grid inventory of inputs and outputs per kWh to make up the use stage inventory per monitor. The effective life scenario models the actual quantity of hours that an average monitor spends in each of the two primary power consumption modes (full-on and a lower power state) during its lifetime. Assumptions used to calculate the kilowatthours per effective life are detailed in Section 2.4. The total electricity consumption for each monitor was calculated to be 634 kWh/effective life (2,282 MJ/effective life) for the CRT and 237 kWh/effective life (853 MJ/effective life) for the LCD.

End-of-life (EOL) Methodology

For the EOL analysis, a monitor is assumed to have reached EOL status when:

- it has served its useful life;
- is no longer functional; and/or
- is rendered unusable due to technological obsolescence.

The major EOL dispositions considered in this analysis are as follows:

- recycling including disassembly and materials recovery;
- landfilling including hazardous [Resource Conservation and Recovery Act (RCRA), Subtitle C] and non-hazardous (RCRA, Subtitle D) landfills;
- remanufacturing including refurbishing or reconditioning (to make usable again); and
- incineration waste to energy incineration.

The functional unit in this analysis is one monitor; therefore, the different EOL dispositions were allocated as a probability of one monitor going to a certain EOL disposition. Data were somewhat scarce on the percent of monitors going to each disposition, especially for LCD monitors, which have not as yet reached EOL. After literature research and consultation with the project's Technical Work Group, as well as various other industry experts, project partners chose best estimates of disposition distributions (Table ES-4).

Disposition	CRT	LCD
Incineration	15%	15%
Recycling	11%	15%
Remanufacturing	3%	15%
Hazardous waste landfill	46%	5%
Solid waste landfill	25%	50%

 Table ES-4. Distribution of EOL disposition assumptions for the CRT and LCD

Sources: NSC, 1999; EPA, 1998; CIA, 1997; EIA, 1999; Vorhees, 2000; TORNRC, 2001

Primary data were collected for CRT recycling from three companies. The data from these companies represent facility operations ranging from October 1999 to February 2000. In the absence of actual data for LCD recycling, data on a CRT shredding-and-materials-recovery process was used to model LCD recycling.

Hazardous/solid waste landfilling and incineration were developed from secondary data obtained from *Ecobilan*. Although data specific to landfilling and incineration operations for monitors alone were not available, existing inventories were available for landfilling and incinerating the following major monitor materials (by weight): steel, glass, plastic, and aluminum. These inventories were combined, based on the approximate proportion of each material in a CRT and an LCD, to create individual processes for landfilling and for incineration (for each monitor type). The majority of the assembled monitors by weight is accounted for in the overall incineration and landfilling processes.

Remanufacturing data were excluded from the assessment because no single set of operations could be identified to adequately represent remanufacturing activities that could be incorporated in our model.

LCI Limitations and Uncertainties

Several factors contribute to the overall quality of data for each life-cycle stage. For example, the manufacturing stage includes several different processes that were collected from several different companies. The quality of one data set from one company may be very different from that of another company. Relative data quality estimates have been made for each life-cycle stage, including electricity generation, which is included in the results of more than one life-cycle stage (Table ES-7). The table also lists the major limitations associated with each life-cycle stage.

Life-cycle stage	Relative data quality	Major limitations		
Upstream	Moderate	Used only secondary data, which has undetermined quality and not originally collected for the purpose of the CDP.		
Manufacturing	Moderate to high	A few data points remain in question.		
Use	Moderate to high	Assumptions regarding use patterns were made.		
EOL	Low to moderate	Used only secondary data for incineration and landfilling processes; no data available for remanufacturing process.		
Electricity generation	High	Used secondary data, however it was collected and modeled for the CDP, resulting in a higher quality rating despite use of secondary data.		

Table ES-7. Relative data quality and major limitations

Although the manufacturing stage was rated in Table ES-7 as having moderate to high data quality, some of the few data points that remained in question had large effects on the results and are therefore described below. Of the data collected from manufacturers, several attempts were made to verify or eliminate outliers in the data; however, uncertainty in some data remained due to large data ranges and outliers. Specific data with the greatest uncertainty include: (1) LCD glass manufacturing data; (2) CRT and LCD glass manufacturing energy inputs; (3) the distribution and amount of fuel/electricity inputs for LCD module manufacturing; and (4) the use of a large amount of liquified natural gas (LNG) as an "ancillary material" in LCD module manufacturing and not as a fuel.

Uncertainties in the LCD glass manufacturing data stem from the fact that no LCD glass manufacturers were willing to supply inventory data. Therefore, the LCD glass manufacturing inventory was derived from the CRT glass manufacturing data modified to exclude leaded compounds from the inventory. Thus, the baseline analysis in this study assumes the energy use per kilogram of CRT glass and LCD glass are equivalent, which is uncertain. In addition to the uncertainty in the difference between energy used to manufacture CRT glass and LCD glass, the energy reported to produce a kilogram of CRT glass varied greatly between the three data sets received for this project, with the highest total energy value being about 150 times that of the smallest value. Due to this large discrepancy and because there were not enough data sets to evaluate the data for outliers, the glass energy data were evaluated in a sensitivity analysis. The high glass energy use values were mostly a function of liquefied petroleum gas (LPG) used as a fuel.

Other data for which large ranges were reported, and which could be important to the results, are energy data from LCD panel/module manufacturing. Energy data provided by six LCD panel/module manufacturers were highly variable in both the distribution of energy sources and the total energy required to produce one LCD panel. The percent of energy from electricity ranged from approximately 3% to 87%, and the total energy per panel ranged from 440 MJ to 7,000 MJ. The average energy use per panel was approximately 2,270 MJ, and the standard deviation was about 2,910 MJ.

Given the wide variability in the data and large standard deviation, CDP researchers evaluated these data for outliers. One data set was found to be a minor outlier and another was found to be a major outlier. These outliers were excluded from the averages used in the baseline analysis, but included in the averages used in a sensitivity analysis (see Section 2.7.3.3).

Finally, a large amount of LNG (194 kg, on average) was reported to be used as an ancillary material (not a fuel) in LCD panel/module manufacturing. CDP researchers confirmed this application of LNG and the amount with the company providing the data, but it is still uncertain due to problems in communication (e.g., the language barrier). This data point remained in the inventory data set for LCD manufacturing, and was assumed to indeed be an ancillary material, and not a fuel. Keeping the LNG ancillary material in the inventory will not affect the energy impact results, since LNG used as an ancillary material is only linked to the production of that material, and not to the use of it as a fuel.

Baseline LCI Results

Tables ES-5 and ES-6 present the total quantity of inputs and outputs for each life-cycle stage of the CRT and LCD based on input and output types.

Table ES-5. CKT inventory by ine-cycle stage						
Inventory type	Upstream	Mfg	Use	EOL	Total	Units ^a
Inputs						
Primary materials	1.58e+01	4.21e+02	2.19e+02	-3.32e+00	6.53e+02	kg
Ancillary materials	2.11e+00	3.54e+00	3.47e+00	1.07e+01	1.98e+01	kg
Water	5.54e+02	1.14e+04	1.14e+03	-2.73e+01	1.31e+04	kg (or L)
Fuels	8.00e+00	4.28e+02	0	-2.95e+00	4.33e+02	kg
Electricity	7.32e+01	1.29e+02	2.29e+03	2.29e-01	2.49e+03	MJ ^b
Total energy	3.66e+02	1.83e+04	2.29e+03	-1.28e+02	2.08e+04	MJ ^b
Outputs						
Air pollutants	3.00e+01	1.83e+02	4.49e+02	2.47e+00	6.64e+02	kg
Wastewater	1.70e+01	1.51e+03	0	-3.65e+00	1.52e+03	kg (or L)
Water pollutants	8.12e-01	2.01e+01	7.02e-02	-6.18e-02	2.09e+01	kg
Hazardous waste	4.89e+02	1.13e+02	0	8.28e+00	9.46e+00	kg
Solid waste	9.55e+00	8.12e+01	8.33e+01	-1.66e+00	1.72e+02	kg
Radioactive waste	4.39e-04	1.80e-04	2.28e-03	2.29e-07	2.90e-03	kg
Radioactivity	3.80e+07	3.78e+06	4.80e+07	4.80e+03	8.98e+07	Bq

Table FS-5 CRT inventory by life-cycle stage

^a Per functional unit (i.e., one CRT monitor over its effective life).

^b 3.6 MJ = 1 kWh

			• •	•		
Inventory type	Upstream	Mfg	Use	EOL	Total	Units ^a
Inputs						
Primary materials	2.35e+02	4.92e+01	8.01e+01	-2.19e+00	3.62e+02	kg
Ancillary materials	1.06e+00	2.04e+02	1.29e+00	2.11e+00	2.08e+02	kg
Water	2.63e+02	2.15e+03	4.25e+02	-1.80e+01	2.82e+03	kg
Fuels	1.47e+01	2.58e+01	0	-1.95e+00	3.86e+01	kg (or L)
Electricity	3.46e+01	3.16e+02	8.53e+02	1.62e-01	1.20e+03	MJ ^b
Total energy	6.33e+02	1.44e+03	8.53e+02	-8.44e+01	2.84e+03	MJ ^b
Outputs						
Air pollutants	1.12e+02	6.48e+01	1.68e+02	1.30e+00	3.46e+02	kg
Wastewater	8.57e+00	3.12e+03	0	-2.41e+00	3.13e+03	kg
Water pollutants	4.60e-01	1.23e+00	2.62e-02	-4.09e-02	1.68e+00	kg (or L)
Hazardous waste	6.72e-03	4.64e+00	0	1.64e+00	6.29e+00	kg
Solid waste	1.31e+01	1.26e+01	3.11e+01	-4.42e+00	5.23e+01	kg
Radioactive waste	2.21e+01	3.14e+03	3.11e+01	-5.23e+00	3.19e+03	kg
Radioactivity	1.20e+07	1.02e+07	1.79e+07	3.40e+03	4.01e+07	Bq

^a Per functional unit (i.e., one LCD monitor over its effective life).
^b 3.6 MJ = 1 kWh

The total inventory results for life-cycle inputs reveal that more primary materials,¹ water, fuels, electricity, and total energy (i.e., fuel energy plus electricity) are used throughout the CRT life-cycle, while more ancillary materials are used throughout the LCD life-cycle. For the life-cycle outputs, the CRT releases more air emissions; water pollutants; hazardous, solid, and radioactive waste; and radioactivity than the LCD. The LCD releases more total wastewater than the CRT. Complete inventory tables for each input and output type by life-cycle stage for the CRT and LCD are provided in Appendix J.

For the CRT (Table ES-5), of the inputs measured in mass, the water inputs in the manufacturing life-cycle stage constitute the majority of the inputs for the entire life cycle. Water inputs from the LPG production process constitute almost 80% of the water inputs for all life-cycle stages. In this inventory, the LPG is used in large quantities as a fuel in CRT glass manufacturing. When considering which life-cycle stage contributes most to an inventory category, the manufacturing stage has the largest inventory by mass for primary materials, ancillary materials, water inputs, and fuel inputs. This is also due to the production of LPG as needed for CRT glass production. Fuel inputs are dominated by the manufacturing stage and electricity inputs are dominated by the use stage. The total energy (which is calculated by converting the mass of the fuel into units of energy and combining the fuel energy with the electrical energy) is dominated by the manufacturing life-cycle stage, again mostly due to the large LPG fuel energy used in CRT glass production.

CRT outputs measured in mass include air emissions, wastewater, water pollutants, and hazardous, solid, and radioactive waste. Wastewater, by mass (or volume), constitutes the greatest output; however, total wastewater volume is not used to calculate water-related impacts. Instead, individual water pollutants are used to calculate water-related impacts. Of the remaining outputs measured in mass, which are used to calculate impacts (i.e., air emissions, and hazardous, solid and radioactive waste), air emissions are the greatest contributor to outputs in mass. Note that radioactivity is measured in Bequerels (Bq) and cannot be compared on the same scale.

Considering each CRT inventory type and their contributions by life-cycle stage, the mass of wastewater and water pollutants are greatest in the manufacturing life-cycle stage (again due to LPG consumption). The outputs of air emissions, hazardous waste, solid waste, radioactive waste, and radioactivity all have the greatest contribution from the use stage.

For the CRT outputs, all the totals represented in Table ES-5 include outputs to all dispositions. For example, water outputs sent offsite to treatment as well as those directly discharged to surface waters are all included. Similarly, hazardous, solid and radioactive waste outputs may be landfilled, treated, or recycled. The inventory shows these as totals; however, when impacts are calculated, the dispositions dictate which inventory items will be used to calculate impacts (Chapter 3).

For the LCD (Table ES-6), of the inputs measured in mass, the water inputs constitute the majority of the inputs for the entire life cycle, and most of the water inputs are in the manufacturing life-cycle stage. When considering which life-cycle stage contributes most to an inventory category, the manufacturing stage has the largest inventory by mass for ancillary materials, fuels, and water inputs. Primary material inputs are dominated by the upstream stages,

¹ Note that the total mass of primary materials includes the inputs to each process, which may duplicate materials used in processes subsequent to other processes. For example, the primary materials used in steel production are added to the steel used as a primary material for monitor assembly.

while electricity inputs are dominated by the use stage. The total energy is dominated by the manufacturing life-cycle stage. Note that LPG production from glass manufacturing does not dominate much of the LCD inventory as it did for the CRT, because of the smaller amount of glass used in the LCD compared to the CRT.

Of the LCD outputs measured in mass (air emissions, wastewater, water pollutants and hazardous, solid, and radioactive waste), wastewater constitutes the greatest output; however, total wastewater volume alone is not used to calculate impacts. Of the remaining outputs measured in mass, which are used to calculate impacts (i.e., air emissions, water pollutants, and hazardous, solid and radioactive waste), air emissions are the greatest contributor to the outputs. Note again, as mentioned for the CRT, that radioactivity is measured in Bequerels (Bq) and cannot be compared on the same scale.

Considering each LCD output type and their contributions by life-cycle stage, the mass of water pollutants is greatest in the manufacturing life-cycle stage, due to the fuel production processes that support fuel consumption in the manufacturing processes being included in the manufacturing life-cycle stage. Wastewater and hazardous waste outputs are greatest in the manufacturing stage; air emissions, solid waste, radioactive waste, and radioactivity have the greatest contribution from the use stage. As with the CRT, all the output totals represented in Table ES-6 include outputs to all dispositions.

III. LIFE-CYCLE IMPACT ASSESSMENT (LCIA)

LCIA Methodology

LCIA involves the translation of the environmental burdens identified in the LCI into environmental impacts. LCIA does not seek to determine actual impacts, but rather to link the data gathered from the LCI to impact categories and to quantify the relative magnitude of contribution to the impact category (Fava *et al.*, 1993; Barnthouse *et al.*, 1997). Further, impacts in different impact categories are generally calculated based on differing scales and therefore cannot be directly compared.

Within LCA, the LCI is a well established methodology; however, LCIA methods are less well defined and continue to evolve (Barnthouse *et al.*, 1997; Fava *et al.*, 1993). For toxicity impacts in particular, there are some methods being applied in practice (e.g., toxicity potentials, critical volume, and direct valuation) (Guinee *et al.*, 1996; ILSI, 1996; Curran, 1996), while others are in development. However, there is currently no general consensus among the LCA community as to one method over another.

The UT LCIA methodology employed in this study calculates life-cycle impact category indicators for a number of traditional impact categories, such as global warming, stratospheric ozone depletion, photochemical smog, and energy consumption. Furthermore, the method calculates relative category indicators for potential chronic human health, aquatic ecotoxicity, and terrestrial ecotoxicity impacts in order to address project partner's interest in human and ecological toxicity and to fill a common gap in LCIAs.

LCIAs generally classify the consumption and loading data from the inventory stage into various impact categories (know as "classification"). "Characterization" methods are then used to quantify the magnitude of the contribution that loading or consumption could have in producing the associated impact. The impact categories included in the CDP LCIA are as

follows: renewable resource use, nonrenewable materials use/depletion, energy use, solid waste landfill use, hazardous waste landfill use, radioactive waste landfill use, global warming, stratospheric ozone depletion, photochemical smog, acidification, air quality (particulate matter loading), water eutrophication (nutrient enrichment), water quality (biological oxygen demand [BOD] and total suspended solids [TSS]), radioactivity, chronic human health effects (occupational and public), aesthetic impacts (odor), aquatic ecotoxicity, and terrestrial ecotoxicity.

Classification of an inventory item into impact categories depends on whether the inventory item is an input or output, what the disposition of the output is, and in some cases the material properties of the inventory item. Outputs with direct release dispositions are classified into impact categories for which impacts will be calculated in the characterization phase of the LCIA. Outputs sent to treatment or recycle/reuse are considered inputs to treatment or recycle/reuse processes and impacts are not calculated until direct releases from these processes occur. Once impact categories for each inventory item are classified, life-cycle impact category indicators are quantitatively estimated through the characterization step.

The characterization step of LCIA includes the conversion and aggregation of LCI results to common units within an impact category. Different assessment tools are used to quantify the magnitude of potential impacts, depending on the impact category. Three types of approaches are used in the characterization method for the CDP:

- **Loading** An impact score is based on the inventory amount (e.g., resource use).
- **Equivalency** An impact score is based on the inventory amount weighed by a certain effect, equivalent to a reference chemical [e.g., global warming impacts relative to carbon dioxide (CO_2)].
 - *Full equivalency* all substances are addressed in a unified, technical model.
 - *Partial equivalency* a subset of substances can be converted into equivalency factors.
- Scoring of inherent properties An impact score is based on the inventory amount weighed by a score representing a certain effect for a specific material (e.g., toxicity impacts are weighed using a toxicity scoring method).

The scoring of inherent properties method is employed for the human and ecological toxicity impact categories, based on the CHEMS-1 method described by Swanson *et al.* (1997). The scoring method provides a hazard value (HV) for each potentially toxic material, which is then multiplied by the inventory amount to calculate the toxicity impact score.

Using the various approaches, the UT LCIA method calculates impact scores for each inventory item for each applicable impact category. Impact scores are therefore based on either a direct measure of the inventory amount or some modification (e.g., equivalency or scoring) of that amount based on the potential effect the inventory item may have on a particular impact category. The specific calculation methods for each impact category are detailed in Chapter 3. Impact scores are then aggregated within each impact category to calculate the various life-cycle impact category indicators.

General LCIA Methodology Limitations and Uncertainties

The purpose of an LCIA is to evaluate the *relative potential* impacts of a product system for various impact categories. There is no intent to measure the *actual* impacts or provide spatial or temporal relationships linking the inventory to specific impacts. The LCIA is intended to provide a screening-level evaluation of impacts. In addition to lacking temporal or spatial relationships and providing only relative impacts, LCA is also limited by the availability and quality of the inventory data. Data collection can be very time consuming and expensive. Confidentiality issues may also inhibit the availability of primary data.

Uncertainties are inherent in each parameter used to calculate impacts. For example, toxicity data require extrapolations from animals to humans and from high to low doses (for chronic effects) and can have a high degree of uncertainty.

Uncertainties also are inherent in chemical ranking and scoring systems, such as the scoring of inherent properties approach used for human health and ecotoxicity effects. In particular, systems that do not consider the fate and transport of chemicals in the environment can contribute to misclassifications of chemicals with respect to risk. Also, uncertainty is introduced where it was assumed that all chronic endpoints are equivalent, which is likely not the case. The human health and ecotoxicity impact characterization methods presented here are screening tools that cannot substitute for more detailed risk characterization methods. However, it should be noted that in LCA, chemical toxicity is often not considered at all. This methodology is an attempt to consider chemical toxicity where it is often ignored.

Uncertainty in the inventory data depends on the responses to the data collection questionnaires and other limitations identified during inventory data collection. These uncertainties are carried into impact assessment. In this LCA, there was uncertainty in the inventory data, which included but was not limited to the following:

- missing individual inventory items,
- missing processes or sets of data,
- measurement uncertainty,
- estimation uncertainty,
- allocation uncertainty/working with aggregated data, and
- unspeciated chemical data.

The goal definition and scoping process helped reduce the uncertainty from missing data, although it is certain that some missing data still exist. As far as possible, the remaining uncertainties were reduced primarily through quality assurance/quality control measures (e.g., performing systematic double-checks of all calculations on manipulated data).

Baseline LCIA Results

Table ES-8 presents the baseline CRT and LCD LCIA indicator results for each impact category. Appendix M presents complete LCIA results by material, process, and life-cycle stage. The indicator results presented in Table ES-8 are the result of the characterization step of LCIA methodology where LCI results are converted to common units and aggregated within an impact

category. Note that the impact category indicator results are in a number of different units and therefore can not be summed or compared across impact categories.

As shown in the table, under the baseline conditions the CRT indicators are greater than the LCD indicators in the following categories: renewable resource use, nonrenewable resource use, energy use, solid waste landfill use, hazardous waste landfill use, radioactive waste landfill use, global warming, ozone depletion, photochemical smog, acidification, air particulates, biological oxygen demand (BOD), total suspended solids (TSS), radioactivity, chronic public health effects, chronic occupational health effects, aesthetics, and terrestrial toxicity. The LCD indicators are greater than the CRT indicators in the following categories: water eutrophication and aquatic toxicity. In addition, as noted in Table ES-8, if phased-out substances are removed from the CRT and LCD inventories, the LCD ozone depletion indicator would exceed that of the CRT. Details of each impact category and major contributors to the impacts in those categories are presented in Chapter 3.

Summary of Top Contributors by Impact Category

Tables ES-9 and ES-10 summarize the top contributors to CRT and LCD life-cycle impacts by impact category. As shown in Table ES-9, CRT impacts are largely driven by two factors: (1) the large amount of LPG fuel used in CRT glass/frit manufacturing, and (2) the relatively large amount of electricity consumed during the use stage. The LPG production process yields the CRT's top contributor in eight of 20 impact categories. Most of this LPG is used as a fuel source in CRT glass manufacturing in the glass/frit process group, which, in turn, produces the top contributor to two of 20 impact categories. Thus, LPG used in the glass/frit process group (primarily CRT glass manufacturing) is ultimately the key driver for CRT impacts in ten categories. Similarly, outputs from electricity generation during the use stage result in the

Table LS-0. Das	Table ES-8. Basenne me-cycle impact category mulcators					
Impact category	Units per monitor	CRT	LCD			
Renewable resource use	kg	1.31E+04	2.80E+03			
Nonrenewable resource use	kg	6.68E+02	3.64E+02			
Energy use	MJ	2.08E+04	2.84E+03			
Solid waste landfill use	m ³	1.67E-01	5.43E-02			
Hazardous waste landfill use	m ³	1.68E-02	3.61E-03			
Radioactive waste landfill use	m ³	1.81E-04	9.22E-05			
Global warming	kg-CO ₂ equivalents	6.95E+02	5.93E+02			
Ozone depletion	kg-CFC-11 equivalents	2.05E-05 ^{b,c}	1.37E-05 ^b			
Photochemical smog	kg-ethene equivalents	1.71E-01	1.41E-01			
Acidification	kg-SO ₂ equivalents	5.25E+00	2.96E+00			
Air particulates	kg	3.01E-01	1.15E-01			
Water eutrophication	kg-phosphate equivalents	4.82E-02	4.96E-02			
Water quality, BOD	kg	1.95E-01	2.83E-02			
Water quality, TSS	kg	8.74E-01	6.15E-02			
Radioactivity	Bq	3.85E+07 ^d	1.22E+07 ^d			

Table FC 0 Descling life and interest and a second in the stand

Tuble LS of Dusenne me cycle impact cutegory indicators					
Impact category	Units per monitor	CRT	LCD		
Chronic health effects, occupational	tox-kg	9.34E+02	6.96E+02		
Chronic health effects, public	tox-kg	1.98E+03	9.02E+02		
Aesthetics (odor)	m ³	7.58E+06	5.04E+06		
Aquatic toxicity	tox-kg	2.25E-01	5.19E+00		
Terrestrial toxicity	tox-kg	1.97E+03	8.94E+02		

Table ES-8. Baseline life-cycle impact category indicators^a

^a Bold indicates the larger value within an impact category when comparing the CRT and LCD.

^b Several of the substances included in this category were phased out of production by January 1, 1996. Excluding phased out substances decreases the CRT ozone depletion indicator to 1.09E-05 kg CFC-11 equivalents per monitor and the LCD ozone depletion indicator to 1.18E-05 kg CFC-11 equivalents per monitor. These ozone depletion indicators are probably more representative of the CDP temporal boundaries and current operating practices. See section 3.3.6 for details.

^c Although the CRT indicator appears larger than the LCD indicator, uncertainties in the inventory make it difficult to determine which monitor has the greater value. Therefore, this value is not shown in bold.

^d Radioactivity impacts are being driven by radioactive releases from nuclear fuel reprocessing in France, which are included in the electricity data in some of the upstream, materials processing data sets. See section 3.3.12 for details.

top contributor to seven CRT impact categories. Note that in 14 of the 20 impact categories, the top contributor to CRT impacts is responsible for more than 50% of impacts.

LCD impacts are not as dominated by a few data points, but a few processes (LCD monitor/module manufacturing and electricity generation in the use stage) are responsible for a large percent of the impacts. As shown in Table ES-10, both of these processes result in the top contributors to six LCD impact categories each. In addition, the process to produce LNG used as an ancillary material in LCD monitor/module manufacturing is the top contributor to an additional impact category (photochemical smog). Note that in 11 of the 20 impact categories, the top contributor to LCD impacts is responsible for more than 50% of impacts.

As a number of the impact results for both monitor types, and for the CRT in particular, are being driven by a few data points with relatively high uncertainty, sensitivity analyses of the baseline results were also conducted.

Impact category	Top contributors				
	Life-cycle stage	Process group	Material	Contribution to impact score	
Renewable resource use	Manufacturing	LPG production	water	79%	
Nonrenewable resource use	Manufacturing	LPG production	Petroleum (in ground)	56%	
Energy use	Manufacturing	CRT glass/frit mfg.	Liquefied petroleum gas	72%	
Solid waste landfill use	Use	U.S. electric grid	Coal waste	38%	

Table ES-9. Summary of top contributors to CRT impacts by impact category

Impact category	Top contributors			
	Life-cycle stage	Process group	Material	Contribution to impact score
Hazardous waste landfill use	End-of-life	CRT landfilling	EOL CRT monitor, landfilled	91%
Radioactive waste landfill use	Use	U.S. electric grid	Low-level radioactive waste	61%
Global warming	Use	U.S. electric grid	Carbon dioxide	64%
Ozone depletion	Use	U.S. electric grid	Bromomethane	49%
Photochemical smog	Manufacturing	LPG production	Hydrocarbons, unspeciated	36%
Acidification	Use	U.S. electric grid	Sulfur dioxide	47%
Air particulates	Manufacturing	LPG production	PM	43%
Water eutrophication	Manufacturing	LPG production	COD	72%
Water quality, BOD	Manufacturing	LPG production	BOD	96%
Water quality, TSS	Manufacturing	LPG production	Suspended solids	97%
Radioactivity	Materials Processing	Steel production, cold- rolled, semi-finished	Plutonium-241 (isotope)	62%
Chronic health effects, occupational	Manufacturing	CRT glass/frit manufacturing	Liquefied petroleum gas	78%
Chronic health effects, public	Use	U.S. electric grid	Sulfur dioxide	83%
Aesthetics (odor)	Manufacturing	LPG production	Hydrogen sulfide	94%
Aquatic toxicity	Manufacturing	CRT tube manufacturing	Phosphorus (yellow or white)	26%
Terrestrial toxicity	Use	U.S. electric grid	Sulfur dioxide	83%

Table ES-9. Summary of top contributors to CRT impacts by impact category

 Table ES-10.
 Summary of top contributors to LCD impacts by impact category

Impact category	tors	-		
	Life-cycle stage	Process group Material		Contribution to impact score
Renewable resource use	Manufacturing	LCD monitor/module mfg.	Water	38%
Nonrenewable resource use	Materials processing	Natural gas production	Natural gas (in ground)	65%
Energy use	Use	LCD monitor use	Electricity	30%
Solid waste landfill use	Use	U.S. electric grid	Coal waste	44%
Hazardous waste landfill use	End-of-life	LCD landfilling	EOL LCD monitor, landfilled	97%
Radioactive waste landfill use	Use	U.S. electric grid	Low-level radioactive waste	44%

Impact category	Top contributors					
	Life-cycle stage	Process group	Material	Contribution to impact score		
Global warming	Manufacturing	LCD monitor/module mfg.	Sulfur hexafluoride	29%		
Ozone depletion	Manufacturing	LCD panel components manufacturing	HCFC-225cb	34%		
Photochemical smog	Materials processing	Natural gas production	Nonmethane hydrocarbons, unspeciated	45%		
Acidification	Use	U.S. electric grid	Sulfur dioxide	31%		
Air particulates	Materials processing	Steel production, cold- rolled, semi-finished	PM	45%		
Water eutrophication	Manufacturing	LCD monitor/module mfg.	Nitrogen	67%		
Water quality, BOD	Manufacturing	LCD monitor/module mfg.	BOD	61%		
Water quality, TSS	Manufacturing	LPG production	Suspended solids	66%		
Radioactivity	Materials processing	Steel production, cold- rolled, semi-finished	Plutonium-241 (isotope)	96%		
Chronic health effects, occupational	Manufacturing	LCD monitor/module mfg.	Liquefied natural gas	58%		
Chronic health effects, public	Use	U.S. electric grid Sulfur dioxide		68%		
Aesthetics (odor)	Manufacturing	LPG production	production Hydrogen sulfide			
Aquatic toxicity	Manufacturing	LCD monitor/module mfg.	Phosphorus (yellow or white)	98%		
Terrestrial toxicity	Use	U.S. electric grid	Sulfur dioxide	68%		

Table ES-10.	Summary of tor	contributors to LCD) imnacts by im	nact category
1 abic LD-10.	Summary of top	contributors to LCD	impacts by m	pace category

Sensitivity Analyses

Due to assumptions and uncertainties in this LCA, as in any LCA, the following sensitivity analyses of the baseline results were conducted: use stage manufactured life scenario, modified glass energy assumptions, modified LCD module manufacturing energy assumptions, and modified LCD EOL distribution assumptions. The sensitivity analyses were chosen because they evaluated data with either the greatest uncertainties or with large uncertainty and major contributors to the inventory results. Table ES-11 shows the different sensitivity analyses or scenarios that are considered in the impact assessment results.

Monitor type	Sensitivity analysis scenario
Baseline a	nalyses (for reference)
CRT	Effective life scenario with average glass energy inputs (all glass manufacturing energy data used)
LCD	<u>Effective life scenario</u> with average glass energy inputs (all glass manufacturing energy data used) and outliers in the LCD module manufacturing energy data removed
Sensitivity	y analyses
CRT	<u>Manufactured life scenario</u> same as baseline except lifespan is based on manufactured life instead of effective life, which results in some revised functional equivalency calculations
LCD	<u>Manufactured life scenario</u> same as baseline except lifespan is based on manufactured life, which results in some revised functional equivalency calculations
CRT	Modified glass energy scenario same as baseline except comparatively high glass manufacturing energy inputs are removed
LCD	Modified glass energy scenario same as baseline except comparatively high glass manufacturing energy inputs are removed
LCD	Modified LCD module energy scenario same as baseline except LCD monitor/ module manufacturing energy outliers are included in the average
LCD	Modified LCD EOL scenario same as baseline except LCD EOL dispositions are modified

Table ES-11.	List of	sensitivity	analysis	scenarios

Based on the sensitivity analyses, it appears that CRT life-cycle impacts are highly sensitive to the glass energy data, and less sensitive to the lifespan assumptions (lifespan assumptions greatly affect the magnitude of CRT life-cycle impacts, but they do not greatly affect the distribution of impacts among life-cycle stages). LCD impacts are less sensitive to the glass energy data and in fact are not greatly affected by any of the sensitivity analysis scenarios, except the longer lifespan under the manufactured life scenario.

Sensitivity results are also useful to interested members of the public who may be evaluating the relative impacts of different monitor types and are interested in whether the CRT or LCD has greater life-cycle impacts in any given impact category. Table ES-12 presents the monitor type with greatest impacts by impact category and by scenario. This information helps us determine whether major assumptions (e.g., the monitor lifespan and LCD EOL distribution assumptions) or uncertain data (e.g., glass energy data and LCD monitor manufacturing energy) are driving results. As shown in the table, the modified glass energy scenario is the only scenario that significantly changes from the baseline. Under this scenario, life-cycle impact results in seven categories reverse direction from the baseline assessment, such that the LCD has greater impacts than the CRT. Therefore, under this scenario, a total of nine out of 20 categories are greater for the LCD than the CRT, compared to two out of 20 categories under the baseline scenario. The only other scenario that affects these results is the manufactured life scenario, when impacts in the water eutrophication category are greater for the CRT than the LCD.

Impact category	Monitor type with greatest impacts by scenario				
	Baseline Manu- Modified Modified factured glass energy LCD mo life energy		Modified LCD module energy	Modified LCD EOL distribution ^a	
Renewable resource use	CRT	CRT	CRT	CRT	CRT
Nonrenewable resource use	CRT	CRT	LCD	CRT	CRT
Energy use	CRT	CRT	CRT	CRT	CRT
Solid waste landfill use	CRT	CRT	CRT	CRT	CRT
Hazardous waste landfill use	CRT	CRT	CRT	CRT	CRT
Radioactive waste landfill use	CRT	CRT	CRT	CRT	CRT
Global warming	CRT	CRT	LCD	CRT	CRT
Ozone depletion	b	b	b	b	b
Photochemical smog	CRT	CRT	LCD	CRT	CRT
Acidification	CRT	CRT	CRT	CRT	CRT
Air particulates	CRT	CRT	CRT	CRT	CRT
Water eutrophication	LCD	CRT	LCD	LCD	LCD
Water quality, BOD	CRT	CRT	LCD	CRT	CRT
Water quality, TSS	CRT	CRT	LCD	CRT	CRT
Radioactivity	CRT	CRT	CRT	CRT	CRT
Chronic health effects, occupational	CRT	CRT	LCD	CRT	CRT
Chronic health effects, public	CRT	CRT	CRT	CRT	CRT
Aesthetics (odor)	CRT	CRT	LCD	CRT	CRT
Aquatic toxicity	LCD	LCD	LCD	LCD	LCD
Terrestrial toxicity	CRT	CRT	CRT	CRT	CRT

 Table ES-12.
 Summary of CRT and LCD LCIA results

^a Based on a qualitative evaluation, not quantitative results.

^b CRT impacts are greater than LCD impacts in this category when all data are included in the inventories, including data for substances that have been phased out. However, LCD impacts are greater than CRT impacts when phased out substances are removed from the inventories (see Section 3.3.6).

IV. QUALITATIVE RISK SCREENING OF SELECTED CHEMICALS

The scope of the DfE CDP included a streamlined Cleaner Technologies Substitutes Assessment (CTSA) component to perform a qualitative risk screening of specific materials or processes. Traditionally, the DfE Program has conducted CTSAs that perform detailed risk characterizations of alternative chemical processes. The streamlined CTSA for the CDP takes a more detailed look than the LCA at the toxic effects of chemicals used in a process, without conducting a complete risk characterization typical of past CTSAs.

Within the human and environmental health effects impact categories of the LCIA, the input and output amounts are used as surrogates for exposure. The additional CTSA-related analyses are intended to better understand the potential exposures to those materials, during any processes that use those materials, in order to try to better understand potential chemical risks.

Lead, mercury, and liquid crystals were selected by the CDP Core Group for further analysis. These materials were selected for their known or suspected toxicity to humans and the

environment, or because they are of particular interest to industry or the U.S. EPA. The analysis of each material summarized or evaluated the following key areas:

- Use of the materials in computer displays;
- Life-cycle inputs and outputs of the materials from computer displays;
- Life-cycle impacts associated with the material inputs and outputs;
- Potential exposures to the material including occupational, public, and ecological exposures;
- Potential human health effects;
- U.S. environmental regulations for the material; and
- Alternatives to reduce the use of the material in computer displays.

The following are the conclusions drawn from the analyses of lead, mercury, and liquid crystal use in the life cycle of both CRTs and LCDs.

Lead

Lead is found in glass components of CRTs, as well as in electronics components (printed wiring boards and their components) of both CRTs and LCDs. It is also a top priority toxic material at the U.S. EPA and the subject of electronics industry efforts to reduce or eliminate its use. The following conclusions were drawn from a focused look at lead's role in the life cycle of the computer display, and its effects on human health and the environment:

- Due to the much greater quantity of lead in the CRT than the LCD, lead-based life-cycle impacts from the CRT ranged from moderately to significantly greater than those from the LCD in every category, with the exception of solid waste landfill use. The most significant difference was in non-renewable resource consumption, where the CRT consumed over 40 thousand times the mass of non-renewable resources attributable to lead over the course of its life cycle than those consumed by the LCD. Other categories where CRTs had notably greater differences in impacts occurred in hazardous waste landfill use, chronic public health effects, and terrestrial toxicity.
- Contributions of lead-based impacts are small relative to the total life-cycle impacts from other materials in the CRT (e.g., glass, copper wire), with the greatest impacts from lead-based CRT outputs occurring in the categories of non-renewable resources, aquatic toxicity, and chronic public health effects (ranging from 0.1 to 0.2% of the overall impact scores in each category).
- For workers, inhalation is the most likely route of exposure to lead which may result in health concerns. General population exposure to lead is most likely to come from incidental ingestion of lead in the soil, or ingestion of lead brought into the household on workers clothing or on shoes. Studies have discovered potentially high concentrations of lead in households within close proximity to certain facilities that use lead.
- Significant worker exposures to lead have been documented by existing studies of several processes which contribute to the life-cycle of the computer displays (e.g., lead smelting). These exposures have been as high as 90 times the OSHA recommended safety levels for exposure to workers at lead smelters. The resulting occupational chronic health effects to

workers from lead exposure likely have been underestimated by the CDP LCIA
methodology, which uses material inputs, and not outputs, as surrogates for exposure.
Lead and lead compounds pose serious chronic health hazards to humans who may
become over-exposed either in the workplace, or through the ambient environment. Lead
exposure is associated with a range of adverse human health effects, including effects on
the nervous system, reproductive and developmental problems, and cancer. Lead persists
in the environment, but is relatively immobile in water under most surface and
groundwater conditions.

• Alternatives are being developed, such as lead-free solders and glass components, that will potentially minimize the future lead content in both CRTs and LCDs.

Mercury

Mercury is contained within the fluorescent tubes that provide the source of light in the LCD. Mercury is also emitted from some fuel combustion processes, such as coal-fired electricity generation processes, which contribute to the life-cycle impacts of both CRTs and LCDs. EPA's concern with mercury and the potential for exposure during manufacturing and end-of-life processes warranted a more detailed analysis of mercury in the CDP. The following conclusions were drawn from a focused look at mercury's role in the life cycle of the computer display, and its effects on human health and the environment:

- The mercury emitted from the generation of power consumed by the CRT (7.75 mg) exceeds the entire amount of mercury emissions from the LCD, including both the mercury used in LCD backlights (3.99 mg) and the mercury emissions from electricity generation (3.22 mg). Although this was not expected because mercury is used intentionally in an LCD, but not in a CRT, the results are not surprising since mercury emissions from coal-fired power plants are known to be one of the largest anthropogenic sources of mercury in the United States. Because the CRT consumes significantly more electricity in the use stage than the LCD, its use stage emissions of mercury are proportionately higher than those of the LCD.
- Contributions from mercury-based impacts are not significant relative to the total lifecycle impacts from other materials (e.g., glass, copper wire) in the CRT or LCD, with the greatest impacts from mercury-based outputs occurring in the aquatic toxicity category (0.4% for CRTs, 0.01% for LCDs).
- Possible pathways of worker exposure during backlight fabrication include inhalation of mercury vapors, and dermal exposure or ingestion of mercury on skin. The most likely pathway for general population exposure is inhalation of mercury released into the air.
- Exposure data relevant to the manufacturing of mercury backlights were not available, therefore specific conclusions about the potential magnitude of worker exposures could not be made. Occupational chronic health effects to workers from mercury exposures calculated during the impact assessment (3.99e-06 tox-kg for LCD, none for CRT) likely have been underestimated by the CDP LCIA methodology, which uses material inputs as surrogates for exposure.
- Mercury and mercury compounds pose serious chronic health hazards to humans who are exposed. EPA has determined that mercury chloride and methyl mercury are possible

human carcinogens. Mercury poses serious chronic health hazards to humans, affecting the nervous system, brain, and kidneys.

Alternative backlights have been developed that not only eliminate mercury from the light, but also improve on many of the optical characteristics of the displays. Current development is focused on improving the energy efficiency of the alternative lights.

Liquid Crystals

Liquid crystals (LCs) are organic compounds responsible for generating the image in an LCD. LCs are not present in CRTs. The toxicity of the LCs in LCDs has been alluded to in the literature, yet there is very little known about the toxicity of these materials. By including LCs in a more detailed analysis, this section attempted to better characterize any potential hazard and/or potential exposure of LCs from the manufacturing, use, and disposal of LCD monitors. The following conclusions were drawn from a focused look at LC's role in the life cycle of the computer display, and its effects on human health and the environment.

- LCs are combined into mixtures of as many as 20 or more compounds selected from hundreds of potential liquid crystal compounds. Because of the possible variations in mixtures and the sheer number of compounds available, a select number of liquid crystals were used to assess potential human health hazards.
- LCs do not appear to contribute significantly to any of the impact categories defined for this study. The total score for LCD occupational impacts based on potential worker exposure to LCs of 4.18 tox-grams, calculated using default toxicity values, represents less than 0.01% of the total overall chronic occupational health effects impact score of 898 tox-kg for the functional unit of one LCD.
- Impacts were not calculated for LC releases in the CDP LCIA because data regarding LC outputs were not available to the project. LCs are not used to fabricate CRTs and so have no environmental impacts in the CRT life cycle.
- Occupational exposures to LCs during the fabrication of the LCD panels are not expected to be significant. The enclosed nature of the chamber in which the LCDs are assembled, combined with the equipment (e.g., gloves, aprons) worn by workers in a clean room environment, are both expected to act to minimize exposures. Other occupational exposures may exist that have not been identified.
- Toxicological testing by a manufacturer of LC substances and mixtures showed that 95.6% (562 of 588) of the liquid crystals tested displayed no acute toxic potential to humans. Twenty-five of the remaining twenty-six chemicals had the potential to exhibit harmful effects to humans, while the remaining crystal was classified as toxic (EU classification) and thus was discontinued. An EPA review of toxicity data for the confidential LC compounds was unable to identify any relevant toxicity information. Insufficient toxicity data exist to assess the toxicity of specific LC compounds.
- Testing for mutagenic and carcinogenic effects by the supplier showed that 99.9% (614 out of 615) of the liquid crystal compounds tested displayed no mutagenic effects. The remaining chemical that showed mutagenic potential was excluded from further development. Additionally, mutagenicity testing of ten LC substances using mammalian cells showed no suspicion of mutagenic potential.

V. SUMMARY AND CONCLUSIONS

The purpose of the CDP, as stated in Chapter 1, is to provide a scientific baseline of lifecycle environmental impacts of CRTs and LCDs, and to develop a life-cycle model for future analyses. The primary targeted audience is the electronics industry, for whom results may provide insight into improvement opportunities in the life cycle of CRTs and/or LCDs. In addition, the general public may also find results useful when considering environmental impacts of each display type. This report, however, does not include direct comparative assertions or improvement assessments based on the results. Alternatively, results and conclusions are described in terms of the overall LCI versus the LCIA, and details of the impact assessment, including the additional assessments of lead, mercury and liquid crystals, and the sensitivity analyses. Major uncertainties, cost and performance considerations, suggestions for improvement opportunities, and suggestions for further research are also provided.

LCI vs. LCIA

Inventory data provide information on how much material is being consumed in the life cycle (i.e., inputs) and how much material is generated/released (i.e., outputs). The LCI alone, however, does not always translate directly into impact categories that may be of interest. Impacts are sometimes driven by materials other than the top inventory contributors. For example, the top air emission for LCDs is carbon dioxide, however the greatest global warming impact score is from SF_6 in the LCD monitor/module manufacturing process.

Some impact categories associated with ancillary materials and water pollutant inventory types had different outcomes in the LCI versus the LCIA. For example, the three impact categories affected by the ancillary materials inventory had greater *impacts* for the CRT (Table ES-8), although the ancillary material *inventory* had greater amounts of inputs for the LCD (see Tables ES-5 and ES-6). In this case, both primary and ancillary materials contribute to the impact categories, contributing to the differing results.

In addition, the LCD had greater inventory amounts of wastewater outputs than the CRT; however, the impacts related to water releases are in some cases greater for the CRT than the LCD. In the LCIA, the LCD has greater impacts for water eutrophication and aquatic toxicity, but not for the two water quality categories (BOD and TSS), chronic health effects to the public, nor terrestrial toxicity, all of which include water emissions in calculating the impact score. These results show that the inventory results may not directly translate into impact results.

CRT and LCD Baseline LCIA Results

The LCIA results showed that the CRT has greater total life-cycle impact indicators than the LCD in most of the impact categories (see Table ES-8). In the baseline scenario, the CRT has greater impacts than the LCD in all but two impact categories (eutrophication and aquatic toxicity). However, note that for the ozone depletion category, the LCIs for both the CRT and LCD contain data for substances that were phased out of production by 1996 due to their ozone depletion potential. Whether these emissions still occur in countries that were signatories to the Montreal Protocol and its Amendments and Adjustments (such as the United States and Japan) is not known, but considered to be unlikely. When phased-out substances are included in the inventory, the CRT has greater ozone depletion impacts than the LCD. However, if phased-out substances are removed from the inventories, the results are switched, with the LCD having greater impacts.

When considering which life-cycle stage has greater impacts, the LCIA results showed that the manufacturing life-cycle stage dominates impacts for most impact categories for both the CRT and LCD. Table ES-13 lists the number of impact categories with the greatest impacts by life-cycle stage.

A more detailed evaluation of lead, mercury and liquid crystals was completed in Chapter 4. As expected, the CRT, which has lead in the glass, frit, and printed wiring boards (PWBs), had greater impacts from lead than did the LCD, which only has lead in the PWBs. Regarding mercury, there were greater inventories of mercury in the CRT life cycle than in the LCD life cycle, despite the fact that only the LCD has mercury directly in the product. The greater amount of mercury in the CRT life cycle is from the release of mercury and mercury compounds from the generation of electricity. The CRT consumes significantly more electricity in the use stage than the LCD. Liquid crystals are only found in LCDs, and therefore, there are no associated impacts for the CRT. Little conclusive information was available on the liquid crystal materials. A detailed literature search was conducted, however very little data were available on the toxicity of these materials. Based on the limited toxicity data obtained, liquid crystals currently do not appear to be a significant human health or environmental hazard in the LCD life cycle. However, there were insufficient toxicity data available to make a definitive conclusion about LC toxicity.

 Table ES-13. Number of impact categories in each life-cycle stage with greatest impacts among life-cycle stages (baseline scenario)

	# of impact categories with greatest impacts among life-cycle stages				
Monitor type	Upstream	Manufacturing	Use	EOL	
CRT	3	9	6	2	
LCD	3	11	4	2	

CRT Results

For the CRT, many of the impacts were driven by a single material in the inventory. As shown in Table ES-9, in 14 of the 20 impact categories, the top individual contributor to the impacts was responsible for greater than 50 percent of the impacts. This shows that the CRT data are highly sensitive to a few data points. Major conclusions from the CRT LCIA are as follows:

- Energy used in glass manufacturing and associated production of LPG are driving the baseline CRT results (they dominate ten impact categories, including overall life-cycle energy use).
- The large amounts of fuel used as energy sources are driving occupational health effects. Occupational impacts are calculated from inventory input amounts, and therefore there may or may not actually be exposure to these fuels (e.g., they may be contained);

however, the results illustrate the potential for health effects, especially under spill or upset conditions.

- The generation of electricity for the use stage dominates seven impact categories.
- Air emissions of sulfur dioxide from electricity generation (for the use life-cycle stage) drive chronic public health effects, acidification, and terrestrial toxicity impacts. This may be a concern, for example, in areas in nonattainment of regulated levels of sulfur dioxide in the United States.

The use of LPG fuel in glass manufacturing dominated ten impact categories: two directly from the LPG used in glass/frit manufacturing (energy use impacts and chronic occupational health effects) and eight from LPG production (renewable resource use, nonrenewable resource use, photochemical smog, air particulates, water eutrophication, BOD water quality, TSS water quality, and aesthetics). In addition, impacts from the generation of electricity during the use stage dominated seven impact categories: solid waste landfill use, radioactive waste landfill use, global warming, ozone depletion, acidification, chronic public health, and terrestrial toxicity. The CRT tube manufacturing process, which represents the most functionally and physically (by mass) significant component of the CRT monitor, only dominated one impact category (aquatic toxicity). Twenty-six percent of the aquatic toxicity score was from phosphorus outputs from tube manufacturing, while most of the rest were from the materials processing life-cycle stage. The remaining two impact categories (hazardous waste landfill use and radioactivity) had greatest impacts from the landfilling of the assumed hazardous proportion of CRT monitors, and the release of Plutonium-241 in steel production, respectively (Table ES-9). The radioactivity impacts are driven by the radionuclide Pu-241, due to the electric grid inventory included in the steel production secondary data set, which includes nuclear fuel reprocessing.

The large amount of LPG reported for glass manufacturing was originally questioned during the data collection and verification stage of this project. While no compelling reason could justify removing the LPG data in the baseline case, a sensitivity analysis was conducted in which the glass energy data were modified. Other sensitivity analyses were also conducted (i.e., manufactured life, modified LCD monitor manufacturing energy, and modified LCD EOL distributions). However, the only scenario that substantially altered results was the modified glass energy scenario (see Table ES-12). It is likely that the actual energy inputs to the glass manufacturing process is somewhere between the baseline and modified glass scenarios. More information is needed on energy used in glass manufacturing, which is driving CRT baseline results.

The additional analyses for the CRT of lead and mercury also revealed that the use of lead could present health risks, but the CDP method for calculating occupational impacts uses only process *inputs* (not outputs) and may not adequately represent occupational exposures and risks. Further refinement of the occupational impact analysis may be warranted.

Although there is no mercury in the CRT monitor, mercury emissions from electricity generation in the CRT life cycle were greater (in mass) than the mercury used in the LCD. Therefore, to reduce mercury emissions from the CRT life cycle, efforts to reduce electricity consumption could be taken. Additionally, changes to the electric grid could also reduce mercury emissions from the CRT life-cycle.

LCD Results

The LCD impact results were less sensitive to an individual input or output than the CRT results, although in 11 of the 20 impact categories an individual input was still responsible for greater than 50% of the total impacts. In general, the LCD results are less uncertain than the CRT results. This is because most of the CRT results are being driven by either glass input data or data from secondary sources, while LCD impacts are being driven more by data from primary sources. Some results to note are as follows:

- The LCD monitor/module manufacturing process group had greatest impacts in six impact categories (Table 3-58).
- Although the top contributor to the energy impact category was electricity consumed in the use stage (30%), the overall energy impacts were greater from the manufacturing stage than the use stage.
- In the glass energy sensitivity scenario, the use stage had greatest energy impacts, although only by a small margin over the manufacturing stage (see Figure 3-26).
- Sulfur dioxide [emitted from electricity generation in the use stage, and constituting only 0.37% of the air emission inventory (see Table 2-49)] dominates the acidification, chronic public health, and terrestrial toxicity impact categories (Table 3-58). The high public health and terrestrial toxicity scores are due to its low non-cancer toxicity value and resulting high hazard value (HV).
- Sulfur hexafluoride (SF₆) from LCD monitor/module manufacturing was the single greatest contributor to the global warming impact score; however, carbon dioxide from the use stage and the materials processing stage also contributed significantly to the global warming impacts (Table 3-25).
- The glass energy inputs did not directly dominate any impact categories, as they did for the CRT (due to the smaller mass of glass in the LCD); however, LPG production (required for the glass energy fuel) dominated two categories: TSS water quality and aesthetics (Table 3-58).
- LNG as an ancillary inventory material was questionably very large and had greatest impacts in two categories: nonrenewable resource use and photochemical smog (Table 3-58; shown there as "Natural Gas Production" due to that process being used as a surrogate for LNG production).

The additional analyses of lead, mercury, and liquid crystals showed that the LCIA alone is not adequate enough to determine all the potential impacts within the life-cycle of the LCD monitors. Further, the LCIA method in this LCA used only process inputs as surrogates for occupational exposure. If the occupational impacts methodology were refined, outputs into the occupational environment should also be considered.

For mercury, which is found in the backlights of the LCD monitors, there is nearly the same amount of mercury by mass emitted to the air during electricity generation as there is mercury used to make the backlight unit. The mass of mercury input for backlights is only about 20% greater than the mercury air emissions from electricity generation (across all life-cycle stages).

EXECUTIVE SUMMARY

Liquid crystals were also identified by the CDP Core Group as a material for which additional information would be reviewed. The LCIA did not find the liquid crystals to be significant contributors to any impact categories; however, this could partially be due to the lack of information on them. The additional analysis revealed limited information; however, qualitatively, it did not show significant potential risk.

CRT vs. LCD Sensitivity Analysis Results

The only sensitivity analysis to show significant difference in the results was the modified glass energy scenario. In comparing the CRT and LCD, the CRT *baseline* scenario had greater impacts than the LCD in all but two impact categories (eutrophication and aquatic toxicity) and possibly three (ozone depletion). In the *modified glass energy scenario*, nine of the 20 categories had greater impacts from the LCD life-cycle than the CRT. Energy use remained greater for the CRT; however, nonrenewable resource use, global warming, photochemical smog, eutrophication, BOD and TSS water quality, chronic occupational health effects, and aesthetics all reversed such that the LCD had greater impacts than the CRT (Table ES-12). As stated above, it is believed that a more true representation of the monitor life cycles lies somewhere between the baseline and modified glass energy scenario. Further work is recommended in clarifying and refining glass energy input information.

Uncertainties

As with any LCA, it is not uncommon for there to be uncertainty associated with such a large data collection effort. The limitations and uncertainties associated with this LCA and LCAs in general have been discussed elsewhere in the executive summary. Two of the largest sources of uncertainty in this LCA that have a significant effect on the results are as follows:

- *CRT and LCD glass manufacturing energy inputs (from primary data)*: The larger amount of glass used in CRTs than LCDs results in the CRT having greater associated uncertainty than the LCD results.
- Secondary data for upstream and fuel production processes: When any one material is used in the life-cycle of either monitor in large quantities, the impacts associated with the inputs and outputs from the production of that material may become significant. For example, LPG and LNG production were both used in significant enough amounts to influence some impact categories. Therefore, the uncertainty in the secondary data becomes important. This highlights the need for a consistent, national (or international) LCI database that is updated regularly.

Other uncertainties associated with individual data points had less effect on the overall results than the uncertainties mentioned above. For manufacturers interested in conducting improvement assessments, closer review of such uncertainties may be warranted.

Another point that should be recognized in the overall LCA of CRTs and LCDs is that CRTs are a more mature technology than LCDs. Changes in LCD manufacturing processes have likely occurred during the development and publication of this report. Therefore, conclusions must be carefully drawn when evaluating the mature CRT compared to the newer LCD technology.

Cost and Performance Considerations

The focus of this study has been on the environmental effects associated with CRTs and LCDs. The environmental attributes or burdens of a product are not expected to be considered alone when evaluating the marketability and commercial success of a product. The cost and performance of each monitor type are obviously critical components to a company's or consumer's decisions of whether to produce or purchase a product. The report briefly addresses direct retail costs of the monitors and electricity costs associated with the monitors; however, a complete cost analysis, including all direct costs (e.g., material costs) and indirect costs (e.g., environmental costs to society) is beyond the scope of this report.

The average retail price of 1997-2000 model year monitors, collected from the manufacturers who supplied data for this project, is \$541 for the CRT and \$1,450 for the LCD. From these data, the LCD is approximately 2.7 times more costly. More recent data show that prices have come down, and the difference in prices between the CRT and LCD has also been reduced.

The costs from the use stage can be represented by the use stage electricity costs. Based on the average cost of residential and commercial electricity in the United States, and the amount of energy consumed per functional unit in the use stage (baseline scenario), the electricity costs to consumers over the life of the monitors during the use stage are \$48 for the CRT and \$18 for the LCD. In addition, the upstream and manufacturing costs of electricity in the baseline scenario for the CRT are approximately \$1.3/functional unit and \$1.5/functional unit, respectively; for the LCD they are \$0.10/functional unit and \$3.4/functional unit, respectively.

The LCA is defined such that the monitor assessments are performed on a functionally equivalent basis. To the extent possible, data were collected on functionally equivalent monitors. When companies were approached to participate in the study, they were informed of the performance specification parameters within which the study boundaries were defined. Therefore, it is assumed that they meet the specifications as presented in Table ES-1, and that they perform relatively equivalently. From the primary data, the reported brightness of the CRT was less than the LCD, otherwise, they are functionally similar.

Improvement Assessment Opportunities and Targeted Audience Uses of Report

To meet the primary objective of providing the display industry with data to perform improvement assessments, the industry should look at the manufacturing life-cycle stage, while recognizing the influences of the other stages. CRT improvement opportunities could include improved energy efficiency during glass manufacturing and display use, as well as reductions in lead content. LCD improvement opportunities could also include improved energy efficiency, especially during manufacturing. Certain materials, such as SF_6 and its contribution to global warming, may also be of concern and an area to focus on in future improvement assessments.

EXECUTIVE SUMMARY

In addition, any improvement assessment should consider how changes in one life-cycle stage will affect impacts in other stages. For example, in Chapter 4 we saw that the mercury inputs and outputs from the intentional use of mercury in an LCD backlight are less (by mass) than the mercury emissions from the CRT use stage, due to the relative energy usage by the CRT and the emissions of mercury from electricity generation. In this example, we can see that on a pure mass basis, a product's energy efficiency is a key consideration and any changes in manufacturing should consider if it will affect changes in use.

Another objective of this study was to provide an LCA model for future analyses. Companies or individuals who have more current data for the CRT or LCD can apply them to the model presented here. For example, changes in an individual process can be identified and incorporated into the model. The other processes that are not expected to change significantly can be left unchanged, and only limited data would need to be altered. This would reduce the time and resources that would normally be required for a complete analysis.

Finally, those interested in comparing the results of the two monitors can apply their own set of importance weights to each impact category to determine their individual decision. For example, if energy impacts are much more important than aesthetics to a particular person, they can weigh energy more heavily in concluding which monitor may have fewer environmental impacts, while keeping in mind the data limitations and uncertainties, as well as cost and performance considerations.

Suggestions for Future Research

Areas where future research could be conducted to refine and/or continue the use of the results in this study are as follows:

- gather more information on energy use in glass manufacturing;
- develop consistent materials and fuel processing data in a national (or international) LCI database that is updated regularly;
- refine and/or update some of the LCD manufacturing data (e.g., LNG data);
- collect more complete EOL data (e.g., remanufacturing data, and primary data for incineration and landfilling) to determine better representation of the EOL impacts;
- conduct more research on the EOL options for LCDs;
- collect more detailed data on landfilling and other treatment processes, such as water treatment where no impacts were calculated;
- update manufacturing data to meet more recent monitor model years;
- conduct a more focused analysis on selected areas for detailed improvement assessments; and
- evaluate process changes or other alternatives against an "average 1997-2000 model year" to evaluate impacts of changes or improvements over time.
REFERENCES

- Barnthouse, L., J. Fava, K. Humphreys, R. Hunt, L. Laibson, S. Noesen, J. Owens, J. Todd, B. Vigon, K. Weitz, J. Young (Eds.). 1997. *Life-Cycle Impact Assessment: The State-of-the-Art*. Society of Environmental Toxicology and Chemistry, Pensacola, FL.
- CIA (Computer Industry Almanac, Inc.). 1997. Computer Industry Almanac. Arlington Heights, IL.
- Curran, M.A. 1996. Environmental Life Cycle Assessment. McGraw-Hill: New York
- EIA (Energy Information Administration). 1999. 1997 Residential Energy Consumption Survey, Housing Characteristics report. Web site available at: http://www.eia.doe.gov/emeu/consumption/reports.html (under 'Residential Households').
- *Ecobilan.* 1999. DEAM, life-cycle inventory database developed by the *Ecobilan* Group, copyright 1999.
- EPA (Environmental Protection Agency). 1998. "Characterization of Municipal Solid Waste in the United States: 1997 update." Report No. EPA530-R-98-007.
- Fava, J., F. Consoli, R. Denison, K. Dickson, T. Mohin, and B. Vigon (Eds.). 1993. A Conceptual Framework for Life-Cycle Impact Assessment. Society of Environmental Toxicology and Chemistry (SETAC) and SETAC Foundation for Environmental Education. Pensacola, FL.
- Guinee, J., R. Heijungs, L. van Oers, D. van de Meent, T. Vermeire, M. Rikken. 1996. LCA Impact Assessment of Toxic Releases. The Hague, The Netherlands.
- IISI (International Iron and Steel Institute). 2001. World Steel Trends and Statistics. Web site accessed at http://www.worldsteel.org/trends_indicators/figures_3.html, on July 31, 2001.
- Kincaid, L. et al. 1996. Cleaner Technologies Substitutes Assessment A Methodology & Resource Guide. U.S. Environmental Protection Agency, Office of Pollution Prevention and Toxics. EPA 744-R-95-002. Washington, DC.
- NSC (National Safety Council). 1999. Electronic Product Recovery and Recycling Baseline Report: Recycling of Selected Electronic Products in the United States, Washington, D.C. May.

- Swanson, M.B., G.A. Davis, L.E. Kincaid, T.W. Schultz, J.E. Bartmess, S.L. Jones, E.L. George. 1997. "A Screening Method for Ranking and Scoring Chemicals by Potential Human Health and Environmental Impacts." Environmental Toxicology and Chemistry. 16(2): 372-383.
- TORNRC (The Oak Ridge National Recycle Center). 2001. Personal communication with D. McFarland and A. Bradley, TORNRC, and R. Dhingra, University of Tennessee, Center for Clean Products and Clean Technologies. April.
- Vorhees, G. 2000. Personal communication with G. Vorhees, Envirocycle, and M. L. Socolof, University of Tennessee, Center for Clean Products and Clean Technologies. July.

Chapter 1

GOAL DEFINITION AND SCOPE

1.1 BACKGROUND

This report presents the results of a voluntary, cooperative project among the Design for the Environment (DfE) Program in the U.S. Environmental Protection Agency's (EPA) Office of Pollution Prevention and Toxics, the University of Tennessee (UT) Center for Clean Products and Clean Technologies, the electronics industry, and other interested parties to develop a model and assess the life-cycle environmental impacts of flat panel display (FPD) and cathode ray tube (CRT) technologies that can be used for desktop computers. The DfE Computer Display Project (CDP) analysis provides a baseline report and the opportunity to use the model as a stepping stone for further analyses and improvement assessments for these technologies.

EPA's Office of Pollution Prevention and Toxics established the DfE Program in 1992 to encourage businesses to incorporate environmental concerns into their business decisions. DfE industry projects are cooperative, joint efforts with trade associations, businesses, public-interest groups, and academia to assist businesses in specific industries to identify and evaluate more environmentally sound products, processes, and technologies. The DfE CDP partnership consists of members of electronic industry trade associations, computer monitor and component manufacturers, suppliers to the electronics industry, academic institutions, EPA, and a public interest group. The direction and focus of this project was chosen by the project partners.

The DfE CDP uses life-cycle assessment (LCA) as an environmental evaluation tool, which is increasingly being used by industry. LCA can be used to evaluate the environmental effects of a product, process, or activity. An LCA looks at the full life cycle of the product from materials acquisition to manufacturing, use, and final disposition. It is a comprehensive method for evaluating the full environmental consequences of a product system. There are four major components of an LCA study: goal definition and scoping, life-cycle inventory, impact assessment, and improvement assessment. LCAs are generally global and non-site specific.

Under the DfE Program, the Cleaner Technologies Substitutes Assessment (CTSA) methodology (Kincaid *et al.*, 1996) was developed to generate information needed by businesses to make environmentally informed choices and to design for the environment. The CTSA process involves comparative evaluations of substitute technologies, processes, products, or materials. Impact areas that are evaluated include human and ecological risk, energy and natural resource use, performance, and cost.

Both evaluation tools have similar objectives; however, their applications generally differ. A CTSA is more site specific and evaluates actual (predicted) impacts. For example, techniques such as health risk assessment are incorporated into a CTSA. An LCA is more global and generic in nature and generally would not incorporate site-specific parameters when evaluating impacts. The LCA may also use surrogate measures to represent impacts instead of predicting or measuring actual impacts.

This project focuses on the LCA, while including some CTSA-related analyses. It performs the broad analysis of the LCA, which also incorporates many of the CTSA components (e.g., risk, energy impacts, natural resource use) into the impact assessment. The analysis also

assesses more specific impacts for selected materials and acknowledges product cost and performance, typical of a CTSA. Because both methodologies require intensive data gathering efforts and can be extensive undertakings, the scope must be carefully and clearly defined. As only selected materials are evaluated for the CTSA, this project could be considered an LCA with a streamlined CTSA component.

Life-cycle assessment (LCA) is a comprehensive method for evaluating the full environmental consequences of a product system. Another related assessment strategy *is lifecycle design*, which is a systems-oriented approach for designing more ecologically and economically sustainable product systems. It integrates environmental requirements into the earliest stages of design so total impacts caused by product systems can be reduced. Environmental, performance, cost, cultural, and legal requirements are balanced (Curran, 1996). Environmental impacts and health risks caused by product development are intended to be reduced. This is very similar to *design for environment* (DfE) programs where environmental issues are incorporated into a product system design process. DfE and life-cycle design have similar objectives, although their origins differ. DfE was developed as an off-shoot of *design for X*, where *X* could be any number of criteria (e.g., manufacturability, testability, reliability, recyclability). In DfE, *X* is environmental protection and sustainability.

The EPA DfE Program, where *X* is also environmental protection and sustainability, promotes risk reduction, pollution prevention, energy efficiency, and other resource conservation measures through process choices at a facility level. EPA's DfE CTSA process also includes an analysis of performance and cost. Typically, EPA's DfE Program focuses less on the entire life cycle and more on evaluating technology or material substitutes to reduce environmental impacts. This project combines the DfE Program's CTSA process (Kincaid *et al.*, 1996) and the LCA process, and thus resembles a life-cycle design approach.

LCAs evaluate the environmental impacts from each of the following major life-cycle stages:

- Raw materials extraction/acquisition;
- Materials processing;
- Product manufacture;
- Product use, maintenance, and repair; and
- Final disposition/end-of-life.

Figure 1-1 briefly describes each of these stages for a computer display product system. The inputs (e.g., resources and energy) and outputs (e.g., product and waste) within each life-cycle stage, as well as the interaction between each stage (e.g., transportation) are evaluated to determine the environmental impacts.

As defined by the Society of Environmental Toxicology and Chemistry (SETAC), the four major components of an LCA are: (1) goal definition and scoping; (2) inventory analysis; (3) impact assessment; and (4) improvement assessment. More recently, the international standard, ISO 14040: Environmental Management—Lifecycle Assessment—Principles and Framework, has defined the four major components of an LCA as: (1) goal and scope; (2) inventory analysis; (3) impact assessment; and (4) interpretation of results. The SETAC and International Standards Organization (ISO) framework are essentially synonymous with respect to the first three components, but differ somewhat with respect to the fourth component, improvement assessment or life cycle interpretation. Improvement assessment is the systematic evaluation of opportunities for reducing the environmental impacts of a product, process, or activity. Interpretation is the phase of LCA in which the findings from the inventory analyses and the impact assessment are combined together, consistent with the defined goal and scope in order to reach conclusions and recommendations. In this study and project report, the goal and scope are the subject of Chapter 1. The inventory analysis and impact assessment are the subjects of Chapters 2 and 3 respectively; and the improvement assessment or life-cycle interpretation are left to the electronics industry given the results of this study. The life-cycle inventory and impact assessment strategies are briefly described below.



Boundary

Figure 1-1. Life-cycle stages of a computer display

The life-cycle inventory (LCI) involves the quantification of raw material and fuel inputs, and solid, liquid, and gaseous emissions and effluents. Traditional LCIs quantify pollutant categories (e.g., volatile organic compounds [VOCs]) rather than particular chemicals. This project also includes a more detailed evaluation of a few specific chemicals found in computer displays (lead, mercury, and liquid crystals) to enable a more thorough evaluation of risk from chemical exposure. The approach to the LCI in this study involves defining product components, developing a bill of materials (BOM), and collecting inventory data on each life-cycle stage for computer displays. Details of the LCI data gathering activities are provided in Chapter 2.

1.2 INTRODUCTION

The life-cycle impact assessment (LCIA) involves the translation of the environmental burdens identified in the LCI into environmental impacts. LCIA is typically a semi-quantitative process involving characterization of burdens and assessment of their effects on human and ecological health, as well as other effects such as smog formation and global warming. Details of the LCIA methodology and results are presented in Chapter 3. This project has furthered the development of LCIA methodology by including health effect concerns into the LCIA. This study also qualitatively assesses exposure and chemical risk of selected chemicals in the life cycles of the computer displays (Chapter 4).

1.2 INTRODUCTION

Goal and scope definition is the first phase of LCA and is important to the CTSA as well. This phase is important because it determines why the LCA or CTSA is being conducted and its intended use, as well as the system and data categories to be studied.

This chapter presents the goal and scope of the DfE CDP, including its purpose and goals, previous research and market trends, descriptions of the product systems being evaluated, and the boundaries used in this study. It incorporates scoping as it is recommended in both the LCA (e.g., Curran, 1996; Fava *et al.*, 1991; ISO, 1996) and CTSA processes (Kincaid *et al.*, 1996).

1.2.1 Purpose

The purpose of this study is two-fold: (1) to establish a scientific baseline that evaluates the life-cycle environmental impacts of flat panel displays (FPDs) and cathode ray tubes (CRTs) for desktop computers, by combining LCA and CTSA methodologies; and (2) to develop a model that can be used with updated data for future analyses. This study evaluates the newer active matrix liquid crystal display (LCD) and the more traditional CRT technologies. The evaluation considers impacts related to material consumption, energy, air resources, water resources, landfills, human toxicity, and ecological toxicity. This study is designed to provide the electronics industry with information needed to improve the environmental attributes of desktop computer displays. It is intended to provide valuable data not previously published, and an opportunity to use the model developed for this project in future improvement evaluations that consider life-cycle impacts. It will also provide the industry and consumers with valuable information to make environmentally informed decisions regarding display technologies, and enable them to consider the relative environmental merits of a technology along with its performance and cost.

1.2.2 Previous Research

While there has been some work done on the life-cycle environmental impacts of either CRTs or LCDs, there has not been a quantitative LCA addressing both CRTs and LCDs. For example, Microelectronics and Computer Technology Corporation (MCC) published an *Electronics Industry Environmental Roadmap* (1994) that qualitatively discussed environmental issues and priority needs for reducing impacts from computer CRTs and FPDs, but this project did not, nor was it intended to, focus on all aspects of the displays' life cycles.

Some of the environmental impacts of CRTs (e.g., energy use, disposal of lead, and other end-of-life issues such as recycling) have been identified but not quantified in previous work, such as that done by EPA's Common Sense Initiative Computer and Electronics Subcommittee. Atlantic Consulting completed a draft LCA of the personal computer (including a 15" monitor) for the European Union's (EU) Eco-Label program (Atlantic Consulting and IPU, 1998). Further, the New Jersey Institute of Technology conducted an LCA of television CRTs, which have the same technology as the computer monitor CRT.

Studies on the environmental impacts of FPDs are much less prevalent. A University of Michigan master's thesis (Koch, 1996) evaluated the environmental performance of an LCD manufacturer. The thesis did not quantitatively assess environmental impacts from all life-cycle stages, as would be done in an LCA. The EU Eco-Label Program also evaluated a portable computer with a 13.3" LCD screen (Orango AB and Atlantic Consulting, 1999). The scope was limited to energy consumption for the impact assessment, and it provided some inventories for raw materials production. Human and ecological toxicity were excluded from the scope of the portable computer analysis.

1.2.3 Market Trends

At present, computer displays using CRTs dominate worldwide markets. They provide a rich, high-resolution display well suited to a wide range of user requirements. However, CRT displays are bulky, use larger amounts of energy to operate than LCDs, and are associated with disposal concerns due to the presence of lead in the glass. Color CRT monitors contain lead to help shield the users from x-rays (x-ray attenuation) and can, under some circumstances, be classified as a hazardous waste when disposed of. Newer technologies, collectively referred to as FPDs, have captured significant market segments. FPDs exhibit desirable qualities such as reduced size and weight and greater portability. Environmentally, they are expected to consume less energy during use and do not use leaded glass. However, they may consume more energy during manufacturing, contain small amounts of mercury, are more costly, and in the past have had lower resolution and image quality than the CRT. The LCD, first used predominately in notebook computers, is now moving into the desktop computer market. The 1998 worldwide market for desktop computer CRTs and LCDs are presented in Table 1-1. Market projections anticipate that LCDs will capture sizable market share for desktop computer displays.

		Number of displays (thousands of units)		
Technology		1998	2001	
CRT				
	Worldwide	88,600	119,100	
	North America	33,801	42,609	
LCD				
	Worldwide	1,300	14,300	
	North America	229	3,787	

 Table 1-1. Desktop display markets - actual for 1998 and projected for 2001

Sources: DisplaySearch, 2001.

1.2.4 Need for the Project

Given the expected market growth of LCDs for computer displays, the various environmental concerns throughout the life cycle of the computer displays, and the fact that the relative life-cycle environmental impacts of LCDs and CRTs have not been scientifically established to date, there is a need for a quantitative environmental life-cycle analysis of desktop computer display technologies. Manufacturers can use these results or the model developed here to identify areas for improvement concerning the environmental burdens. Further, as companies or consumers are considering investing in certain displays, they can refer to the results of this study to assist them in making environmentally informed decisions.

1.2.5 Targeted Audience and Use of the Study

The electronics industry is expected to be one of the primary users of the study results. The study is intended to provide industry with an analysis that evaluates the life-cycle environmental impacts of selected computer display technologies. Scientific verification of the relative environmental impacts will allow industry to consider environmental concerns, along with traditionally evaluated parameters of cost and performance, and to potentially redirect efforts towards products and processes that reduce releases of toxic chemicals and reduce risks to health and the environment. Given the results, the industry can then perform an improvement assessment of the display technologies. This also allows the electronics industry to make environmentally informed choices about display technologies when assessing and implementing improvements such as changes in product, process, and activity design; raw material use; industrial processing; consumer use; and waste management.

Another result of the study is an accounting of the relative environmental impacts of various components of the computer displays, thus identifying opportunities for product improvements to reduce potential adverse environmental impacts and costs. Identification of impacts from the computer display technologies can also encourage industry to implement pollution prevention options, such as development and demonstration projects, and technical assistance and training. Since this study incorporates a more detailed health effects component than in traditional LCAs, the electronics industry can use the tools and data to evaluate the health, environmental, and energy implications of the technologies. With this evaluation, the U.S. electronics industry may be more prepared to meet the demands of extended product responsibility that are growing in popularity in the global marketplace, and better able to meet competitive challenges in the world market. In addition, the results and model in this study will provide a baseline LCA upon which alternative technologies can be evaluated. This will allow for more expedited display-related LCA studies, which are growing in popularity by industry and may be demanded by original equipment manufacturers (OEMs) or international organizations.

EPA and interested members of the public can also benefit from the results of the project. The project has provided a forum for industry and public stakeholders to work cooperatively, and the results can be used by stakeholders as a scientific reference for the evaluated display technologies. The results of the project could also be of value to other industries involved in designing environmental improvements into the life cycle of consumer products.

1.3 PRODUCT SYSTEM

1.3.1 Functional Unit

The product system being analyzed in this study is a standard desktop computer display that functions as a graphical interface between computer processing units and users. The desktop market was chosen to evaluate CRTs and FPDs because it represents a large market for CRTs and an anticipated large market for some FPDs. Also, there are a limited number of technologies that meet desktop specifications. Therefore, focusing on the desktop market will affect a significant number of products, while keeping the scope of this project manageable.

The product system is the computer display itself and does not include the central processing unit (CPU) of the computer that sends signals to operate the display. It is assumed that the LCDs operate with an analog interface, and therefore are compatible with current CRT CPUs as plug-and-play alternatives.

In an LCA, product systems are evaluated on a functionally equivalent basis. The functional unit is used as the basis for the inventory and impact assessment to provide a reference to which the inputs and outputs are related. For this project, the functional unit is one desktop computer display over its lifespan. Data collected in this project have been normalized to a display that meets the functional unit specifications, which are presented in Table 1-2. These product performance specifications are assumed to meet the requirements of the system functional unit in the predictable future (i.e., computer technology as predicted through the year 2001). The CRT technology is the current industry standard for this product system.

Specification	Measure
display size ^a	17" (CRT); 15" (LCD)
diagonal viewing area ^a	15.9" (CRT); 15" (LCD)
viewing area dimensions	12.8" x 9.5" (122 in ²) (CRT); 12" x 9" (108 in ²) (LCD)
resolution	1024 x 768 color pixels
brightness	200 cd/m ²
contrast ratio	100:1
color	262,000 colors

 Table 1-2.
 Functional unit specifications

^a An LCD is manufactured such that its nearest equivalent to the 17" CRT display is the 15" LCD. This is because the viewing area of a 17" CRT is about 15.9 inches and the viewing area of a 15" LCD is 15 inches. LCDs are not manufactured to be exactly equivalent to the viewing area of the CRT.

Besides the CRT display, several FPD technologies were considered for inclusion in this study. Among the FPD technologies that exist, the amorphous silicon (a:Si) thin-film transistor-(TFT) active matrix LCD technology meets the requirements of the functional unit within the parameters of this analysis and is assessed in this study. Section 1.3.3.1 describes the LCD technology further, and Appendix A briefly describes several FPD technologies and explains why the non-LCD technologies are not considered for standard desktop computer uses as defined for this study. The following subsections briefly describe both CRT and LCD technologies.

1.3.2 Cathode Ray Tube

1.3.2.1 CRT technology

CRT monitors are a mature technology and are the current industry standard for desktop computer displays. The technology is the same as that for a television. CRT displays use high voltages to accelerate electrons toward a luminescent material (phosphor) that is deposited on a faceplate. The phosphor converts the kinetic energy of the electrons into light. In color CRTs, the phosphors are patterned in dots or stripes of red, green, and blue phosphors. The electrons are emitted from three cathodes in an electron gun assembly and pass through an apertured metal "shadow mask" during their passage to the phosphor. Electrons from each cathode that are directed at the wrong color phosphor are absorbed by the shadow mask. Pictures are created by first focusing the electron beams into tiny spots, which are moved by deflecting the electron beams electron gun assembly efficient in that it only requires three video drivers and connections instead of the 2000 or so in the most common FPD (MCC, 1994).

The high voltages used to accelerate the electrons must be insulated from the external surfaces of the tube and the CRT must have excellent electrical insulating properties. The decelerating electrons produce x-rays and the CRT must also be a good x-ray absorber. Leaded glass surrounds the cathode ray tube to absorb the x-rays.

The major parts of the CRT display are the faceplate (glass panel), shadow mask (also referred to as the aperture mask), a leaded glass funnel, and the electron gun with the deflection yoke. Various connectors, wiring, an implosion band, printed wiring boards (PWBs), and the casing comprise most of the rest of the display. Table 1-3 presents a more comprehensive list of the CRT components and a list of the component materials identified from disassembling a monitor, and additional research to identify the material makeup of some components (MCC, 1998).

Component parts					Materials
		Phosphor-coated faceplate	Glass panel (faceplate)	▶	Glass (1-2.5% PbO alkali/alkaline earth aluminosilicate)
			Phosphors	▶	ZnS, Y ₂ O ₂ S (powders): Sn, Si, K, Cd
			Photoresist	▶	Polyvinyl alcohol
	Faceplate assembly		Black matrix coating (grille dag)	▶	Aquadag**
			Lacquer coating	▶	Mixture of alcohol and plastics
			Aluminum coating	▶	Al
		Internal electron shield *	·		Al
		Shadow mask assembly	Mask		Steel, Ni
			Supports	>	CrNiFe and NiFe
	Frit (lead solder glass)			·>	Lead oxide, zinc borate (~70% PbO)
Tube		Glass funnel		>	Leaded glass (~24% PbO)
	Conductively coated funnel	Conductive coating	┨▶		Aquadag** (may also add iron oxide)
		Frit	 	····· ►	PbO, zinc oxide, boron oxide
		Binder	 	▶	Nitrocellulose binder, amyl acetate
		Neck glass		·>	Leaded glass (30% PbO alkali/alkaline earth silicate)
		Deflection yoke]	▶	Cu, ferrite
	Neck	Base & top neck, rings]	▶	Polystyrene
		Brass ring, brackets]	▶	Brass
		Rubber gaskets]	▶	Rubber
		Screws, washers]	▶	Zn-plated steel
		Neck clamp]	▶	Steel
		Insulating rings	┫		Polysulphone
		Neck PWB	┫		Misc. electronics and resin board
	Implosion band			•	Steel

 Table 1-3. Preliminary list of CRT display components and materials

	Component parts	Materials	
	Electrostatic field shaping electrodes	>	300/400 series steels (Fe, Ni, Cr)
	Cathodes	▶	Ni (coated with mixtures of Ba, Sr, CaCO3)
	Glass pillars	▶	Borosilicate glass
Electron gun	Wire heater (filament heater)	▶	Tungsten, aluminum oxide
	Glass stem	▶	Leaded glass (29% PbO alkali aluminosilicate)
	Springs and washers		Steel
	Gunmount (glass)	▶	Potassium aluminosilicate sintering
Powerboard	Main CRT PWB (includes power supply PWB, etc.)		Misc. electronics and resin board
	Flyback transformer	►	Misc.: e.g., potting material (epoxy), steel, Cu
Casing (chassis and base)			Polystyrene
	Brackets	>	Brass
	Brackets	►	Polystyrene
	XY controls	▶	Polycarbonate
	Connectors	▶	Al
Other misc.	Screws	►	Zn-plated steel
parts	Brackets, anode connection	▶	Q. 1
	Shields (right, left, top, back)		Steel
	Rubber feet	┨▶	Silicone rubber
	Anode cap	┨▶	Rubber
	Insulator pad	▶	Polyester

 Table 1-3. Preliminary list of CRT display components and materials

1.3.2.2 CRT manufacturing

The manufacturing process of the CRT (Figure 1-2) involves first preparing the glass panel and shadow mask. The shadow mask is a steel panel with a mask pattern applied through photolithography. The color phosphors and a black matrix coating (aquadag) are applied to the faceplate, also using photolithography, which involves several steps and several chemicals that etch the material into specific patterns. A lacquer coating is applied to the phosphor-coated glass to smooth and seal the inside surface of the screen, and an aluminum coating is added to enhance brightness and as a conductor to allow the use of voltages over 12 kilovolts without charging of the phosphor screen. An electron shield (typically aluminum) is attached to the shadow mask assembly to prevent stray electrons from reaching outside the screen area (EPA, 1995; MCC, 1994). This then comprises the faceplate assembly. The two major remaining parts are the funnel and electron gun.



Figure 1-2. Traditional twisted nematic AMLCD manufacturing process

The leaded-glass funnel is washed and coated with a black coating (aquadag), which is a good electrical conductor and a non-reflective surface. The funnel and faceplate assembly are joined using frit (solder glass) that is approximately 70% lead oxide. The electron gun is an assembly of glass and several metal parts that are heated to embed the metal in the glass. The electron gun is then fused to the funnel neck. Finishing steps are conducted to complete the manufacturing of the CRT (or "tube") (e.g., attaching a metal implosion band for implosion protection and safety) and then the entire monitor is assembled with other necessary parts (e.g., the main and neck PWBs, power cord, casing).

1.3.3 Liquid Crystal Display

LCDs are used for various applications, with their largest market currently in notebook computers. LCDs have been gaining a presence in the desktop computer display market and are expected to continue to do so. Compared to the CRT, the LCD provides a more compact display, as well as being of higher contrast, sunlight readable, more reliable, and more durable (i.e., requiring much less maintenance) (Koch and Keoleian, 1995). In general, the major functional disadvantages have been that the resolution and quality of the image have not matched that of CRTs. However, emerging technologies are expected to meet user requirements.

The two most common types of LCDs are passive matrix (PMLCD) and active matrix (AMLCD). Brief descriptions of these and other LCD technologies are presented in Appendix A, Table A-1. In 1998, AMLCDs constituted approximately 81% and PMLCDs 19% of computer monitor LCD production (MCC, 1998). PMLCDs are used primarily for low-end products (e.g., cannot perform video applications); AMLCDs are used for high-end multimedia products and better meet the specifications for standard desktop computers. PMLCDs are forecasted to decline to less than one percent of the LCD desktop display market by 2002 (Young, 1998). Therefore, this project's focus is on AMLCDs.

1.3.3.1 LCD technology

In general, an LCD is comprised of two glass plates surrounding a liquid crystal material that filters external light. LCDs control the color and brightness of each pixel (picture element) individually, rather than from one source, such as the electron gun in the CRT.

The most common type of AMLCD, and the one that meets the functional unit specifications of this project, is the a:Si TFT AMLCD (see Appendix A, Table A-1 for descriptions of various types of AMLCDs). AMLCDs consist of driver tabs along the columns and rows of the display glass and transparent parallel electrical lines across the glass arranged to form a matrix. Each intersection of the matrix forms a pixel. The TFT AMLCD has a transistor at each pixel which functions as an electronic switch to activate an individual pixel. This active addressing technique allows for high contrast between the on and off states of a pixel.

Operation of the AMLCD is determined by how the liquid crystals are aligned when activated by an electrical current. Traditional AMLCDs use a twisted nematic (TN) operating principle of the liquid crystal, which is evaluated in this study. The orientation of the liquid crystal molecules either allows or does not allow light from a backlight source to pass through the display cell. When no electrical current is present, the liquid crystals align themselves parallel to a polyimide orientation film (alignment layer) on the glass. When a current is applied, the liquid

crystals turn perpendicular to the glass. The combination of the alignment layer, electrical charge, and polarizers that are laminated to the glass panels affect an on or off state of the LCD cell (see Appendix B for further explanation; also see Castellano [1992] and OTA [1995]). The backlight supplies the light source for the display and generally has four cold cathode fluorescent tubes that contain small amounts of mercury vapor. Because the LCD technology essentially regulates passage of a backlight through the display, LCDs are considered non-emitting display technologies. CRTs, on the other hand, are emitting displays which emit electrons to illuminate appropriate phosphors.

There is a variation of the a:Si TFT AMLCD technology called in-plane switching (IPS). Compared to the traditional TN mode, IPS TFT allows for a wider viewing angle that is comparable to the CRT. The traditional TN mechanism vertically twists the liquid crystals by sending voltage through the display from electrodes that are on both the front and rear glass panels. IPS mode twists the liquid crystals horizontally in response to a voltage applied by electrodes on the rear glass only. The TN TFT uses indium-tin oxide (ITO) as the transparent electrode on the front and rear glass panels. For the IPS TFT, no ITO is needed on the front glass and the electrode on the rear glass is made of any of a number of other materials (e.g., Mo, Ta, Al/Cr, MoW). Therefore, no ITO is used for the IPS mode. However, the IPS TFT display demands an increase in the number of backlights to meet the brightness requirements for desktop applications (DisplaySearch, 1998). Although this technology may be produced for displays that meet this study's functional unit, the manufacturers of 15" AMLCDs that supplied data for this study provided data only for traditional a:Si TFT AMLCDs. IPS was forecasted to have a 35% market share of LCD desktop monitors in 2000 (Young, 1998) and therefore, studying IPS technology may be an area for future research.

Based on the disassembly of an LCD monitor conducted by MCC, a summary of the materials that are in an AMLCD are presented in Table 1-4 (MCC, 1998). The major components of the AMLCD are the glass panel (which includes the transistors, electrodes, liquid crystals, orientation film, polarizers, and row and column drivers), the backlight assembly (including the cold cathode fluorescent tube, light guide, and associated electronics), other electronics (main LCD controller), and the stand and cover. The remaining components and materials are listed in Table 1-4. In this report we will also refer to the LCD "module" as a component. This is comprised of the LCD panel, backlight unit, and main LCD controller. Module manufacturing as a process modeled in this study includes the major process in LCD manufacturing, which is panel manufacturing described below.

LCD component parts					Materials
			Glass	>	Soda lime or borosilicate
			Thin film transistor (TFT)	▶	Misc. (e.g., Si, Mo, Al, etc) *
			Electrode	▶	Indium-tin oxide (ITO)
		AMLCD cell	Polarizers	▶	Iodine, cellulose triacetate-acrylic,
	I CD glass papel				etc.
	assembly		Orientation film (alignment layer)	•	Polyimide
			Liquid crystals	>	e.g., phenylcyclohexanes biphenyls
			Color filters	>	Resins
LCD assembly		Row/column driver tabs		▶	IC chip on polyimide
		Row/column driver PWBs		▶	Misc. (Si, Cu, etc.)
		Connection flex]	▶	Cu on film of polyimide
	Plastic frame			►	Polycarbonate, glass filled
	Gaskets]		▶	Silicone rubber
	Screws			►	Steel (Fe)
	Metal clip]			BeCu
		->			
	Brightness enhancer]		►	Polyester
	Cable assembly	Misc. wires & connectors		·•	Misc. (Cu, plastic, etc.)
Controller		J			Misc. (Si, Cu, etc.)
(PWB)	- >				
	Housing			▶	Steel (Fe)
D	Screws			►	
Power supply assembly	Insulator			►	Polyester
	Power switch			►	
	Power cord recepticle			▶	ABS/Cu
	Heat sink			▶	Aluminum
	Power supply PWB	 		▶	Misc. (Si, Cu, etc.)

 Table 1-4. LCD components and materials

		LCD component parts		Materials
	Metal plate Screws			Steel (Fe)
	Brass threaded stand off			Brass
	Gasket			Foam rubber
	Nylon strain relief			
	Nylon clamp			Nylon
	Clear protector			Plexiglas
	Opaque diffuser			
D	White reflector		>	Polyester
Backlight assembly	Light pipe			Polycarbonate
	Corner tape			Aluminized mylar
	Light assembly	Cold cathode tube		Glass, phosphor, Hg
		Shock cushion	┨▶ Г	Silicone rubber
	İ	Cable assembly	┨▶ Г	Insulated Cu
		Rear plate		
	Rear plate assembly	Screws	▶	Steel (Fe)
		Hold-down plate	┨▶	
		Cable clamp	┨▶ Г	Nylon
		Plastic tube	┨▶ Г	Polycarbonate
		Flat cable toroid	┨▶ Г	Hi-mu ferric
		Caution label	┨▶ Г	Paper
	Backlight PWB		>	Misc. (Si, Cu, etc.)

 Table 1-4. LCD components and materials

		LCD component parts	Materials
	Screws		
	Metal plate	▶	Steel (Fe)
Rear cover	BeCu fingers		BeCu
assembly	Cloth mesh		
	Insulator		Polyester
	Rear cover	▶	Plastic (ABS)
	Brackets & washers		
	Axle & spring		Steel (Fe)
	Base weight		
Base/stand	C-clip		
assembly	Swivel bearing 1	▶	Stainless steel
	Swivel bearing 2	▶	Polyoxymethylene (acetal)
	Covers		Plastic (ABS)
	Upright		unsaturated polyester, glass filled
	Bushing		Nylon
	Rubber feet		Silicone rubber
	Screws		
	Power supply cover		Steel (Fe)
	Front bezel		
Other misc. parts	Knob		ABS
	Power switch		
	LED light pipe	▶	Polycarbonate
	Adjustment PWB	┃▶	Misc. (Si, Cu, etc.)
	Cable clamp	┃▶	Nylon
	Insulator	▶	Polyester

 Table 1-4. LCD components and materials

Source: MCC, 1998.

*Example of TFT materials: gate metal (Al or Cr), S_iO₂, SiN, a-Si/SiNx, a-Si, drain metal

1.3.3.2 LCD manufacturing

LCD technology uses a glass substrate (e.g., soda lime or borosilicate glass). Once the glass is acquired, it must be cleaned, which is a critical step in reliable manufacturing. The manufacturing techniques of the LCD are similar to the production of semiconductor chips, which require energy intensive "clean rooms" for manufacturing. A conductor or semiconductor is deposited on the glass substrate. Most FPDs require a transparent conductor (electrode) such as ITO, which is usually deposited by a sputtering method. TFT devices in AMLCDs use semiconducting materials (e.g., silicon, cadmium selenide) for transistors at each pixel. These semiconducting materials can be prepared by vacuum evaporation, using either electron beam evaporation, sputtering, or chemical vapor deposition. Electrode patterns are then formed by a photolithographic process that begins with the coating of photoresist on the ITO or metal-coated substrate.

The photoresist (photosensitive organic polymer) is then "developed" using liquid organic chemicals. This is similar to the manufacture of silicon integrated circuits (ICs). The ITO or metal is then etched to form electrodes. FPD manufacturing employs etchants (e.g., H_3PO_4/HNO_3 , HF/HCl, HClO₄, CF₄/SF₆) that differ from those used in silicon IC manufacturing. LCDs use alignment layers and iodine/dye based polarizers. AMLCDs use amorphous silicon, silicon nitride, various oxides and metals. The silicon IC manufacturing process of plasma-enhanced chemical vapor deposition (PECVD) is used for some of these layers, as are more conventional vacuum deposition techniques. Finally, various organic liquids (the liquid crystals) are injected between the top and bottom substrate of LCDs (MCC, 1993).

The general process flow of AMLCD manufacturing is presented in Figure 1-3. The photolithographic subprocess steps for the front and rear glass panels are presented in Figures 1-4 and 1-5. Various processes include sputtering, PECVD, photolithography, wet etching, reactive ion etching (RIE), in-process testing, liquid crystal processing, and lamination among others. The manufacturing process of IPS differs from traditional TN in that there are fewer photolithography steps required since no ITO is patterned onto the front or rear glass plates.

Once the photolithographic processes are completed, the polyamide orientation film is applied and rubbed on the glass. The front and rear glass substrates are merged and the liquid crystals are inserted. Polarizers are laminated to the front and rear panels, completing the "AMLCD cell" of the LCD glass panel assembly (see Table 1-4). The electronic components that operate the AMLCD cell (i.e., the column and row driver tabs and wiring boards) are then attached to the glass. This completes the LCD glass panel assembly. The other major components of the LCD are then assembled. Adding the controller PWB and the backlight assembly make up what may also be called the "LCD module." Finally, the power supply assembly and the plastic cover and stand are added to make an assembled monitor.

In total, there are six PWBs needed for the LCD: controller, row driver, column driver, backlight, power supply, and adjustment knob PWB. The major ones by size and function are the larger controller and backlight PWBs, and the smaller column and row PWBs. These are compared to the two major PWBs in the CRT: the main and neck PWBs (see Table 1-3).



Figure 1-3. Traditional twisted nematic AMLCD manufacturing process



Figure 1-4. Front glass AMLCD manufacturing (photolithography)



Figure 1-5. Rear glass AMLCD manufacturing (photolithography)

1.4 ASSESSMENT BOUNDARIES

1.4.1 Life-Cycle Stages and Unit Processes

In a comprehensive cradle-to-grave analysis, the display system includes five life-cycle stages: (1) raw materials acquisition; (2) materials processing; (3) product manufacture; (4) product use, maintenance and repair; and (5) final disposition/end-of-life. Also included are the activities that are required to affect movement between the stages (e.g., transportation). The major processes within the life cycles of CRTs and LCDs, which are modeled in this study, are depicted in Figures 1-6 and 1-7, respectively. Details on collecting data for these processes are presented in Chapter 2.

1.4.2 Spatial and Temporal Boundaries

The geographic boundaries of this assessment are shown in Table 1-5. This LCA will focus on the U.S. display market; therefore, the geographic boundary for the use and disposition stages of displays is limited to the United States. Raw materials acquisition and material processing for materials used in the manufacture of computer displays are done throughout the world. Product manufacturing is done predominately in Asia, although there are foreign-owned desktop display manufacturers operating plants in the United States and other countries. Therefore, for purposes of this study, the geographic boundaries for raw material extraction, material processing, and product manufacture are worldwide.

Life-cycle stage	Geographic coverage	
Raw materials acquisition	worldwide ^a	
Material processing	worldwide ^a	
Product manufacture	worldwide ^a	
Use	United States	
Disposition	United States	

 Table 1-5. Geographic coverage for each life-cycle stage

^a In this study, worldwide boundaries were considered; however, the actual geographic locations for LC1 data are presented in Chapter 2 (Sections 2.2 and 2.3).

While the geographic boundaries show where impacts might occur for various life-cycle stages, traditional LCAs do not provide an actual spatial relationship of impacts. That is, particular impacts cannot be attributed to a specific location. Rather, impacts are generally presented on a global or regional scale.



Note: Arrows indicate flows between life-cycle stages. The arrows also represent transportation of materials required to get from one process to another.

* Electricity generation and fuel production (fuel oils, natural gas, and LPG) processes are not shown but were attached to those processes that consume energy resourcesthroughout the life-cycle

Figure 1-6. CRT Life-Cycle



Note: Arrows indicate flows between life-cycle stages. The arrows also represent transportation of materials required to get from one process to another. * Electricity generation and fuel production (fuel oils, natural gas, and LPG) processes are not shown but were attached to those processes that consume energy resourcesthroughout the life-cycle



This study addresses impacts from the life cycle of a desktop computer display manufactured using 1997-2000 technology. The use and disposition stages cover a period that represents the life of a display. Two lifespans are considered: (1) the "effective" life, defined as the period of time the display is in use by primary, secondary, or even tertiary users before reaching its final disposition; and (2) the "manufactured" life, defined as the designed durability of a display. The effective life is estimated based on past and current use patterns of displays and represents a realistic estimate of the lifespan. Because the effective life is subject to many variables, including fluctuating market trends, it is also necessary to evaluate the displays over their manufactured life. The manufactured life is estimated based on the manufacturer's estimated durability of the display. Because of quickly changing technologies in this industry, the effective life has been shorter than the manufactured life. The effective life, which is currently the more realistic scenario is used as the primary scenario in the final results presented in this study. However, the manufactured life is presented as an alternative scenario.

It is assumed that parameters that may change with time, such as available landfill space, will remain constant throughout the lifespan of the product system. If the lifespan is relatively short (i.e., within a timeframe where significant changes in landfill space would not occur), the preceding assumption is reasonable. If resources become more scarce within the lifespan, this assumption could underestimate the impacts.

1.4.3 General Exclusions

Impacts from the infrastructure needed to support the manufacturing facilities (e.g., maintenance of manufacturing plants) are beyond the scope of this study. However, maintenance of clean rooms used in the manufacturing of LCDs (and other components), which require substantial amounts of energy, are considered part of the manufacturing process.

REFERENCES

Atlantic Consulting and IPU. 1998. LCA Study of the Product Group Personal Computers in the EU Ecolabel Scheme. LCA Study Version 1.11. January.

Castellano, J. 1992. Handbook of Display Technology. Stanford Resources, Inc., San Jose, CA.

- Curran, M.A. 1996. Environmental Life-Cycle Assessment. McGraw-Hill, New York, NY.
- DisplaySearch. 2001. Personal communications with R. Young, DisplaySearch, and J.G. Overly, University of Tennessee, Center for Clean Products and Clean Technologies. July.
- DisplaySearch. 1998. DisplaySearch FPD Equipment and Materials Analysis and Forecast. Austin, TX.
- EPA (Environmental Protection Agency). 1995. *EPA Office of Compliance Sector Notebook Project - Profile of the Electronics and Computer Industry*. EPA310-R-95-002. Office of Enforcement and Compliance Assurance. Washington, DC. September.
- Fava, J., R. Denison, R. Jones, B. Curran, M. Vigon, B. Selke, S. & J.A. Barnum. 1991. *Technical Framework for Life-Cycle Assessment*. Society of Environmental Toxicology and Chemistry & SETAC Foundation for Environmental Education, Inc. Washington, DC.
- FRC (Fuji Chimera Research). 1996. "The Future of Liquid Crystal and Related Display Materials." Originally published in Japan by FRC and translated by InterLingua Linguistic Services, Inc. Redondo Beach, CA. 1997.
- ISO (International Standards Organization). 1996. ISO 14040, Environmental Management -Life-cycle Assessment Principles and Framework. TC 2071 SC 5N 77. International Standards Organization, Paris.
- Kincaid, L. et al. 1996. Cleaner Technologies Substitutes Assessment A Methodology & Resource Guide. U.S. Environmental Protection Agency, Office of Pollution Prevention and Toxics. EPA 744-R-95-002. Washington, DC.
- Koch, J. 1996. Evaluating Environmental Performance: A Case Study in the Flat Panel Display Industry. University of Michigan, Natural Resources and Environment, Ann Arbor, MI. April.
- Koch, J. and G. Keoleian. 1995. "Evaluating Environmental Performance: A Case Study in the Flat-Panel Display Industry." Proceedings of IEEE International Symposium on Electronics and Environment, Orlando, May 1-3, pp. 158-165.

- MCC (Microelectronics and Computer Technology Corporation). 1993. Environmental Consciousness: A Strategic Competitiveness Issue for the Electronics and Computer Industry. Austin, TX.
- MCC. 1994. Environmental Industry Environmental Roadmap. Austin, TX.
- MCC. 1998. *Computer Display Industry and Technology Profile*. EPA 744-R-98-005. December.
- NEC Corporation. 1998. "The LC Monitor Market as Predicted by a Monitor Manufacturer." NEC Home Electronics, Inc. In *Flat Panel Display 1998 Yearbook*. Nikkei Microdevices. Originally published by Nikkei Business Publicationa and translated by InterLingua Linguistic Services, Inc., Redondo Beach, CA. 1997.
- OTA (Office of Technology Assessment). 1995. *Flat Panel Displays in Perspective*. OTA-ITC-631. U.S. Government Printing Office, Washington, DC. September.
- Orango A.B. and Atlantic Consulting. 1999. "EU Eco-Label for Portable Computers-Suggested Criteria." Version 1.12. Götebprg, Sweden. May 19, 1999.
- SRI (Stanford Resources, Inc.). 1998. Web site available at: http://www.stanford.resources.com/sr/main/homepage. April.
- Young, R. 1998. Personal communications with R.Young, DisplaySearch, and M.L. Socolof, University of Tennessee Center for Clean Products and Clean Technologies, June.

this page intentionally left blank

Chapter 2

LIFE-CYCLE INVENTORY

A life-cycle inventory (LCI) is the identification and quantification of the material and resource inputs and emission and product outputs from the unit processes in the life cycle of a product system (Figure 2-1). For the Design for the Environment (DfE) Computer Display Project (CDP), LCI inputs include materials used in the computer display product itself, ancillary materials used in processing and manufacturing of the displays, and energy and other resources consumed in the manufacturing, use, or final disposition of the displays. Outputs include primary products, co-products, air emissions, water effluents, and releases to land. Specific unit processes for CRTs and LCDs are represented by the boxes in Figures 1-6 and 1-7, and each unit process for both the CRT and LCD life cycles, and graphically displays how they are linked to subsequent processes. This figure will be referred to throughout this chapter in the discussion of each life-cycle stage.





This chapter describes the methods for collecting LCI data in the DfE CDP, and presents LCI results. Section 2.1 describes the general methodology for LCI data collection, while Sections 2.2 through 2.6 present the specific methodologies, data sources, data quality, limitations and uncertainties for each life-cycle stage. Section 2.7 then concludes with the combined LCI data for each monitor type.

More specifically, Section 2.2 presents the LCI methodology for the materials extraction and materials processing (i.e., "upstream") life-cycle stages, including electricity generation. Electricity is used in several processes throughout each monitor's life-cycle, and the electricity generating process is linked to the processes that use electricity. As a consequence, the inventory results from electricity generation are reported as part of the associated life-cycle stage for the process to which it is linked. For example, electricity used in manufacturing a product component is included in the manufacturing stage inventory, while electricity used during the use stage is included in the use stage inventory.



Figure 2-2. CRT linked processes



Figure 2-3. LCD linked processes

Section 2.3 presents the LCI methods for the product manufacturing life-cycle inventory, which was developed from primary data collected through questionnaires designed and sent to manufacturers for this study. Sections 2.4 and 2.5 present the methods for developing the use and end-of-life (EOL) life-cycle stage inventories, respectively. Transportation is another important aspect of a product life-cycle that can cause environmental impacts. Information related to the transport of materials, products, and wastes is presented in Section 2.6. As the final section of Chapter 2, Section 2.7, summarizes and discusses the entire inventory data over the life-cycles of the CRT and LCD computer monitors. Sensitivity analyses are also discussed in Section 2.7.

2.1 GENERAL METHODOLOGY

This section describes the data categories evaluated in the CDP LCI, the decision rules used to determine which materials extraction and materials processing life-cycle stages (e.g., "upstream" processes) to evaluate in the study, and data collection methods. It also describes procedures for allocating inputs and outputs from a process to the product of interest (e.g., a display or display component) when the process is used in the manufacture, recycle, or disposal of more than one product type at the same facility. Finally, it describes the data management and analysis software used for the project and methods for maintaining overall data quality and critical review.

2.1.1 Data Categories

Table 2-1 describes the data categories for which inventory data were collected, including material inputs, energy inputs, natural resource inputs, emission outputs, and product outputs. Inventory data were normalized to mass per functional unit (in the case of material and resource inputs and emission or material outputs), megajoules (MJ) per functional unit (in the case of energy inputs), or the number of components per functional unit (in the case of display components). As discussed in Section 1.3, the functional unit is one desktop computer display over its lifespan.

Data that reflected production for one year of continuous processes were scaled to one functional unit. Thus, excessive material or energy associated with startups, shutdowns, and changeovers were assumed to be distributed over time. Consequently, any environmental and exposure modeling associated with the impact assessment reflects continuous emissions such that equilibrium concentrations may be assumed. If the reporting year was less than one year for any inventory item, the analysis was adjusted as appropriate.

Data Category	Description		
Material inputs (kg pe	r functional unit)		
Primary materials	Actual materials that make up the final product for a particular process. These can be individual materials or a combination of materials that comprise a component part.		
Ancillary (process) materials	Materials that are used in the processing of a product for a particular process. Process materials from monitor manufacturing could include, for example, etchants used during photolithography which are washed away and not part of the final product, but are necessary to manufacture the product.		
Energy inputs (MJ per	r functional unit)		
Process energy	Energy consumed by any process in the life-cycle.		
Precombustion energy	The energy expended to extract, process, refine, and deliver a usable fuel for combustion.		
Transportation energy Energy consumed in the transportation of the materials or products in the life			
Natural resource input	ts (kg per functional unit)		
Non-renewable resources	Materials extracted from the ground that are non-renewable, or stock, resources (e.g., coal).		
Renewable resources (e.g., water)	Water or other renewable, or flow, resources (e.g., limestone) are included in the analysis. Renewable resource data values are presented in mass of water consumed for a particular process.		
Emissions outputs (kg	per functional unit)		
Air	Mass of a product or material that is considered a pollutant within each life-cycle stage. Air outputs represent actual gaseous or particulate releases to the environment from a point or diffuse source, after passing through emission control devices, if applicable.		
Water	Mass of a product or material that is considered a pollutant within each life-cycle stage. Water outputs represent actual discharges to either surface or groundwater from point or diffuse sources, after passing through any water treatment devices.		
Solid wastes	Mass of a product or material that is deposited in a landfill or deep well. Represents actual disposal of either solids or liquids that are deposited either before or after treatment (e.g., incineration, composting), recovery, or recycling processes.		
Products (kg of materi	al or number of components per functional unit)		
Primary products	Material or component outputs from a process that are received as input by a subsequent unit process within the display life cycle.		
Co-products	Material outputs from a process that can be used, either with or without further processing, that are not used as part of the final functional unit product.		

Table 2-1.	LCI data	a categories
------------	----------	--------------

2.1 GENERAL METHODOLOGY

Data were also collected on the final disposition of emissions outputs, such as whether outputs are recycled, treated, and/or disposed. This information helps determine what impacts will be calculated for a particular inventory item. Methods for calculating impacts are discussed in Chapter 3, Life-Cycle Impact Assessment. The dispositions used for this project are as follows:

- air,
- surface water,
- landfill,
- land (other than landfill),
- treatment,
- recycling/reuse, and
- deep well injection.

Given the enormous amount of data involved in inventorying all of the inputs and outputs for a product system, life-cycle assessment (LCA) practitioners typically employ decision rules to make the data collection manageable and representative of the product system and its impacts. Section 2.1.2 discusses the decision rules used in the CDP LCI.

2.1.2 Decision Rules

In an LCA, the materials extraction and materials processing life-cycle stages (referred to as "upstream" life-cycle stages) include processes for extracting raw materials from the earth and processing those raw materials into the materials used in the manufacture of the product of interest. Examples of upstream processes include the mining of iron ore and its processing with other materials into steel sheet or the extraction of petroleum from underground reserves and its conversion into plastic pellets. A continuing challenge for LCA practitioners is to collect all the appropriate data for a product system, including data for these upstream processes as well as data for product manufacturing, use, and disposal processes. In this project, decision rules as to what inventories should be included as upstream processes in the overall modeled life-cycle are based on the materials used to manufacture the computer monitors. Also, which component parts to include in the model depends on these decision rules. In considering upstream materials, a combination of several factors, including availability of existing data were considered. For considering which component manufacturing processes to include, the decision rules, plus manufacturers willingness to participate, factored into our overall scope of what was included in the analysis.

To help determine which upstream processes to include in the CDP LCI, first the bill of materials (BOM) of the component parts and the materials that make up those parts (Tables 1-3 and 1-4) was reviewed. The quantities of those materials identified by MCC can be found in Industry Profile Document (MCC, 1998). Using the MCC BOM allowed work to begin on selecting and collecting upstream data before the actual BOM from the manufacturing stage was obtained from the project's primary data collection effort. Final decisions on which upstream processes to include were based on the BOM developed from data collected from manufacturers. Tables 2-2 and 2-3 list the BOMs of the primary material inputs from the manufacturing of the CRT and LCD, respectively. The mass quantities given in the tables are the primary material
inputs to the manufacturing process and would be equivalent to the amount of material in the final product plus excess or waste materials. While this is not exactly equivalent to the mass of each material that makes up a finished product, it does represent the mass of primary materials used to manufacture the finished product at participating facilities. Details of how these data were obtained are presented in Section 2.3, which describes the manufacturing stage inventory.

Material/Component		Mass	(kg) ^a	weight % of total inputs ^a	
	Sub-component				
Lead oxide	glass	9.76		46.1%	
	Lead		0.45		2.1%
Steel		5.16		24.4%	
Plastics		3.04		14.4%	
	Polycarbonate (PC)		0.92		4.36%
	Styrene-butadiene co-polymer		0.83		3.91%
	Polyethylene ether (PEE)		0.74		3.47%
	Acrylonitrile-butadiene-styrene (ABS)		0.32		1.52%
	High-impact polystyrene (HIPS)		0.15		0.71%
	Triphenyl phosphate		0.05		0.25%
	Tricresyl phosphate		0.02		0.11%
	Phosphate ester		0.01		0.04%
Printed wir	ing boards (PWB) and components	0.85		4.00%	
Cables/wir	es	0.45		2.13%	
Aluminum (heat sink)		0.27		1.29%	
Nickel allo	y (invar)	0.27		1.29%	
CRT shield	lassembly	0.24		1.14%	
Ferrite		0.17		0.80%	
Deflection	yoke assembly	0.15		0.71%	
Demagneti	c coil	0.13		0.60%	
Video cabl	e assembly	0.11		0.54%	
Power cord	lassembly	0.11		0.54%	
Electron gu	ın	0.10		0.47%	
CRT magn	et assembly	0.08		0.36%	
Audio cabl	e assembly	0.07		0.34%	
Frit		0.07		0.32%	
Solder		0.03		0.13%	
Phosphors		0.02		0.08%	
Aquadag		0.02		0.07%	
Other (mise	c.)	0.06		0.30%	
TOTAL		21.16		100%	

Table 2-2. Bill of primary material inputs for a 17" CRT monitor

^a Based on the primary material inputs to the manufacturing process, including material in the final product plus excess or waste materials.

Material/Co	Material/Component		Mass (kg) ^a		weight % of total inputs ^a	
	Subcomponent					
Steel		2.53		44.12%		
Plastics		1.78		30.98%		
	Polycarbonate (PC)		0.52		9.00%	
	Poly(methyl methacrylate) (PMMA)		0.45		7.80%	
	Styrene-butadiene copolymer		0.36		6.31%	
	Polyethylene ether (PEE)		0.30		5.23%	
	Triphenyl phosphate		0.09		1.61%	
	Polyethylene terephthalate (PET)		0.06		1.03%	
Glass	Glass			10.31%		
Printed wiring boards (PWB) and components		0.37		6.52%		
Cables/wires		0.23		4.08%		
Aluminum (ł	Aluminum (heat sink, transistor)			2.34%		
Solder (60%	tin, 40% lead)	0.04		0.66%		
Color filter p	pigment	0.04		0.65%		
Polyvinyl alo	cohol (PVA) (for polarizer)	0.01		0.15%		
Liquid crysta	als, for 15" LCD, unspecified ^b	0.0023		0.04%		
Backlight lar	mp (cold cathode fluorescent lamp, CCFL)	0.0019		0.03%		
Mercury			3.99E-06		0.0001%	
Transistor m	Transistor metals, other (e.g., Mo, Ti, MoW)			0.03%		
Indium tin oxide (ITO) (electrode)		0.0005		0.01%		
Polyimide alignment layer		0.0005		0.01%		
Other (e.g., a	adhesives, spacers, misc.)	0.0031		0.05%		
TOTAL		5.73		100%		

 Table 2-3. Bill of primary material inputs for a 15" LCD monitor

^a Based on the primary material inputs to the manufacturing process, including material in the final product plus excess or waste materials.

^b This does not include all liquid crystals, as those specified as individual chemicals are in very small amounts and included in the "other" category.

The decision rule process begins by assessing the materials and components in Tables 2-2 and 2-3 for the following attributes:

- 1. The mass (M) contribution of each component and material in the display. The mass is important in order to account for the majority of materials and components that make up a display, but also because the more significant the material or component by mass, the more materials and resources may be required to manufacture the material or component and thus it may have a significant environmental impact.
- 2. *Materials that are of known or suspected environmental (Env) significance (e.g., toxic).* As this is an environmental life-cycle assessment, consideration of materials or components that are known to or are suspected to exhibit an environmental hazard are also included to the extent feasible.

- 3. *Materials that are known or suspected to have a large energy (E) contribution to the systems energy requirements.* Energy impacts are of great interest to the use and manufacture of display monitors and, therefore, priorities were given to including materials or components that are known to or suspected to consume large amounts of energy.
- 4. *Materials or components that are functionally (F) significant to the display.* "Functionally significant" is defined as important to the technically successful operation of the display. For example, the liquid crystals in an AMLCD would be "functionally significant" while screws, gaskets, or the plastic cover would not be.
- 5. *Materials or components that are physically (P) unique in the CRT as compared to the LCD and vice versa.* The physical uniqueness of a material or component could be identified by chemical makeup or by size. An example of the latter would be if the plastic casing for the CRT and LCDs were made of the same material, but the CRT casing had substantially more material by weight.

The priority scheme depicted in Figure 2-4 provides guidelines for applying the CDP decision rules. Material or component inputs that account for more than five percent of the total mass of a display technology were given top priority for data collection, as were those of known or suspected environmental or energy significance and those that are functionally significant or physically unique. Of less emphasis in trying to obtain data, but still included if possible, were materials or components that are functionally significant but physically similar to those in the other technologies, and those that were between 1 and 5% of the total mass of the display. Recognizing the limitations of project resources, materials or components that account for less than one percent of total mass or are not otherwise significant or unique were excluded *a priori*.

Based on this hierarchy and on review of the preliminary BOMs, Tables 2-4 and 2-5 present how components were rated based on the priority scheme, and which ones were included in the analysis. If a material or component was included in the analysis as a separate process, it is listed in the last column of Tables 2-4 and 2-5 by what life-cycle stage that process is in (i.e., upstream or manufacturing process). If a material or component was only included as part of another process, and not as a separate process in the profile, the process in which that material or component is found is provided in the last column.



Figure 2-4. Decision rule guidelines

*e.g., materials are excluded if they are not of known environmental significance (for example, toxic) or are not physically unique.

Table 2-4.	Decision rule priorities and scope of analysis for the primary material inputs
	of the 17" CRT monitor

Material/Component		Decision rule	Included in analysis as:	
	Sub-component			
Lead oxide	glass	M, F, P, E, Env	manufacturing process	
	Lead	P, Env	upstream process	
Steel		М	upstream process	
Plastics				
	Polycarbonate (PC)	М	upstream process	
	Styrene-butadiene co-polymer	М	upstream process	
	Polyethylene ether (PEE)	М	part of monitor assy. process	
	Acrylonitrile-butadiene-styrene (ABS)	М	upstream process	
	High-impact polystyrene (HIPS)	none	upstream process	
	Triphenyl phosphate	none	part of monitor assy. process	
	Tricresyl phosphate	Р	part of monitor assy. process	
	Phosphate ester	Р	part of monitor assy. process	
Printed wiring boards (PWB) and components		M, E, Env	manufacturing process	
Cables/wire	es	none	part of monitor assy. process	
Aluminum (heat sink)		M (<5%), E	upstream process	
Nickel allo	y (invar)	M (<5%), P	upstream process	
CRT shield	lassembly	none	part of monitor assy. process	
Ferrite		Р	upstream process	
Deflection	yoke assembly	F, P	part of monitor assy. process	
Demagneti	c coil	F, P	part of monitor assy. process	
Video cable	e assembly	none	part of monitor assy. process	
Power cord	lassembly	none	part of monitor assy. process	
Electron gu	in	F, P	part of tube mfg. process	
CRT magn	et assembly	Р	part of monitor assy. process	
Audio cabl	e assembly	none	part of monitor assy. process	
Frit		F, P, E, Env	manufacturing process	
Solder		Env	part of monitor assy. process	
Phosphors		F, P	part of tube mfg. process	
Aquadag		F, P	part of tube mfg. process	
Other (mise	c.)		miscellaneous	

 \mathbf{M} = mass is greater than 1% of the total display weight; \mathbf{Env} = environmental/toxic concern; \mathbf{E} = energy concern; \mathbf{F} = functional (technological) importance; \mathbf{P} = physically unique. Note: The CRT processes included in the CDP LCA were presented in Figures 1-6 and 2-2.

Table 2-5.	Decision rule priorities and scope of analysis for the primary material inputs
	of the 15" LCD monitor

Material/Component		Decision rule	Included in analysis as:
	Sub-component		
Steel		М	upstream process
Plastics			
	Polycarbonate (PC)	М	upstream process
	Poly(methyl methacrylate) (PMMA)	M, P	upstream process
	Styrene-butadiene co-polymer	М	upstream process
	Polyethylene ether (PEE)	М	part of monitor assy. process
	Triphenyl phosphate	M (<5%)	part of monitor assy. process
	Polyethylene terephthalate (PET)	M (<5%), P	upstream process
Glass		М	manufacturing process
Printed wiring boards (PWB) and components		M, E, Env	manufacturing process
Cables/wires			part of monitor assy. process
Aluminum (heat sink, transistor)		M (<5%), E	upstream process
Solder (60%	% tin, 40% lead)	Env	part of monitor assy. & module mfg. processes
Color filter	pigment	Р	part of color filter patterning
Polyvinyl a	lcohol (PVA) (for polarizer)	none	part of polarizer mfg. process
Liquid crys	tals, for 15" LCD	F, P, Env	manufacturing process
Backlight la	amp (cold cathode fluorescent lamp)	F, P, Env	manufacturing process
	Mercury	Env	part of backlight lamp process
Transistor r	netals, other (e.g., Mo, Ti, MoW)	F, P	part of module mfg. process
Indium tin oxide (ITO) (electrode)		F, P	part of module mfg. process
Polyimide a	lignment layer	F, P	part of module mfg. process
Other (e.g.,	adhesives, spacers, misc.)		miscellaneous

 \mathbf{M} = mass is greater than 1% of the total display weight; \mathbf{Env} = environmental/toxic concern; \mathbf{E} = energy concern; \mathbf{F} = functional (technological) importance; \mathbf{P} = physically unique.

Note: The LCD processes included in the CDP LCA were presented in Figures 1-7 and 2-3.

2.1.3 Data Collection and Data Sources

Data were collected from both primary and secondary sources. Primary data are directly accessible, plant-specific, measured, modeled, or estimated data generated for the particular project at hand. Secondary data are from literature sources or other LCAs, but are specific to either a product, material, or process used in the manufacture of the product of interest.

Table 2-6 lists the types of data (primary or secondary) used for each life-cycle stage in the CDP LCI. In general, greater emphasis was placed on collecting data and/or developing models for product manufacturing, use, and end-of-life. Primary data were collected from product and component manufacturers (in the U.S., Japan, and Korea), and CRT recyclers who voluntarily agreed to participate in the project. When proprietary data were involved, the University of Tennessee (UT) Center for Clean Products and Clean Technologies entered into confidentiality agreements with the affected company. In addition, to both protect confidentiality and better represent the various manufacturing processes, data for particular processes that were collected from more than one company, where possible, and aggregated. Attempts were made to get at least two companies to contribute data for each particular process in the manufacturing life-cycle stage, which resulted in some process datasets being the aggregate of multiple (2-7) companies data. However, this was not feasible in every case and some datasets were simply the data of one company. Details of the data aggregation methods are provided in Section 2.3.

Data for the use stage were modeled specifically for this project by UT researchers, but were based on secondary data (i.e., secondary data were built upon to create the data used in the inventory for the use life-cycle stage). Data associated with the electricity generation were also based on secondary data, but modeled for this project. Transportation information (e.g., transportation mode and distances) were collected from the manufacturers that provided primary data. These data were linked to secondary data inventories of fuel inputs and emissions outputs for various types of transport vessels. Transportation data cover movement of materials and components both into and out of a facility, but do not include transportation of packaging or distribution of the finished display to the consumer. Finally, secondary data were used for upstream processes. More details on each of these data collection efforts are provided in subsequent sections of this chapter.

Life-cycle stage	Data types
Upstream (materials extraction and processing)	Secondary data.
Product and component manufacturing	Primary data, except secondary data used for frit.
Use	Modeled using secondary data; maintenance and repair are not included in the analysis.
Final disposition (recycling and/or disposal)	Modeled using secondary data plus primary data from CRT recycling facilities.
Packaging, transportation, distribution	Primary data from product and component manufacturers for transport mode and distance; secondary data for fuel inputs and emissions outputs for the transport vessel. Packaging and distribution not included.

 Table 2-6. Data types by life-cycle stage

In some instances, neither primary nor secondary data were available. For example, CRTs are a much more mature technology than the LCD, and end-of-life (EOL) data are much less prevalent for the LCD than for the CRT. Where primary and secondary data are lacking, various assumptions and modeling serve as defaults.

2.1.4 Allocation Procedures

An allocation procedure is required when a process within a system shares a common management structure, or where multiple products or co-products are produced. In the CDP LCI allocation procedures are used when processes or services associated with the functional unit (e.g., a desktop computer display over its lifetime) are used in more than one product line at the same facility (e.g., notebook computers, televisions). For example, transistors are used in LCD desktop computer displays, but also in other LCD technologies, such as notebook computer displays. If a facility uses a single process to manufacture transistors for both desktop and notebook computer display, inputs and outputs are allocated among the product lines to avoid over-estimating the environmental burdens associated with the product under evaluation.

The International Standards Organization (ISO, 1996) recommends that wherever possible, allocation should be avoided or minimized. This may be achieved by sub-dividing the unit process into two or more sub-processes, some of which can be excluded from the system under study. In the example above, if a manufacturer of transistors supplied only desktop computer LCD manufacturers, no allocation would be necessary from that manufacturer. However, it is more likely that the transistor manufacturer would have a larger customer base, including manufacturers of various products other than liquid crystal desktop computer displays. This requires allocation of flows from the manufacturing of transistors for several products to those associated only with desktop computer displays. As suggested by ISO, if sub-processes within the transistor facility can be identified that distinguish between transistors manufactured for LCDs and for other products, the latter sub-processes can be eliminated from the analysis, thus reducing allocation procedures.

In this study allocation procedures are used as follows:

- Inventory data for utilities and services common to several processes are allocated to reflect the relative use of the service. For example, fuel inputs and emission outputs from electric utility generation are allocated to a display or display component according to the actual or estimated electricity consumed during the manufacture, use, or final disposition of the product. Similarly, fuel inputs and emission outputs from commercial transport of a display component to a display assembler are allocated to the display component according to the mass of the component, the distance traveled, and the fraction of the transport vehicle's capacity occupied by the number of components shipped.
- Where a unit process produces co-products, the burdens associated with the unit process are allocated to the co-product on a mass basis. In the transistor example above, burdens are allocated according to the total mass of transistors used in desktop displays and the mass used in notebook computers. Total mass can be calculated from sales records which document the number of transistors delivered to different customers and measured mass of a set number of transistors.

Allocation is also necessary when a single process produces both energy and products. In this case, the inputs are partitioned among the energy and products, as appropriate, to avoid allocating inapplicable chemical burdens to energy production. However, this scenario was not encountered in the CDP LCI.

2.1.5 Data Management and Analysis Software

The data that were collected for this study were either obtained from questionnaires developed for this project, from existing databases, or from primary or secondary data collected by the UT Center for Clean Products and Clean Technologies. All these data were transferred to spreadsheets, which were then imported into a Life-Cycle Design Software Tool developed by the Center for Clean Products and Clean Technologies with funding from the EPA Office of

Research and Development and Saturn Corporation. The software tool was developed to store and organize life-cycle inventory data and to calculate life-cycle impacts for a product profile. Written using Microsoft FoxPro programming software, the tool is designed to allow flexibility in conducting life-cycle design and life-cycle assessment functions. It provides the means to organize inventory data, investigate alternative scenarios, evaluate impacts, and assess data quality.

The UT Life-Cycle Design Software Tool organizes data in such a way that each process inventory is independent. Customized "profiles" (e.g., the manufacture of a CRT or the whole life-cycle of an LCD) can be developed by linking processes together. The tool has the flexibility to modify or replace any particular process within a profile to evaluate potential alternatives. The data provided in this study may serve as a baseline to compare alternatives or modifications to particular processes. The models developed for the life-cycles of the CRT and LCD in this study can remain useful as many of the individual processes in the CRT and LCD life-cycles will likely remain constant (e.g., steel manufacturing, plastics manufacturing). Changes to specific processes can be made to conduct analyses of current or emerging process or technology changes. Relatively quick life-cycle analyses can be conducted on future product or process improvements, given the baseline data already available through this study.

2.1.6 Data Quality

LCI data quality can be evaluated based on the following data quality indicators (DQIs): (1) the source type (i.e., primary or secondary data sources); (2) the method in which the data were obtained (i.e., measured, calculated, estimated); and (3) the time period for which the data are representative. LCI DQIs are discussed further in *Life-Cycle Assessment Data Quality: A Conceptual Framework* (SETAC, 1994). CDP data quality for each life-cycle stage is discussed in detail in Sections 2.3 through 2.6 and summarized below.

For the primary data collected in this project, participating companies reported the method in which the data were obtained and the time period for which the data are representative. Data from the 1997-2000 time period were sought, with the most recent data preferred. Similarly, the time period of secondary data and method in which the data were originally obtained was also recorded, where available. Secondary data cover a broader time period, with data for most materials from the 1997 to 1998 time period, and data for most fuels from the 1983 to 1993 time period.

Anomalies and missing data are common hurdles in any data collecting exercise. Anomalies are extreme values within a given data set. Any anomaly identified during the course of this project that is germane to project results was highlighted for the project team and investigated to determine its source (e.g., mis-reported values). If the anomaly could be traced to an event inherently related to the process, it was left in the data set. If, however, the anomaly could not be accounted for, it was removed from the data set. Specific anomalies highlighted by the project team are discussed in Section 2.7, Summary of Life-Cycle Inventory Results.

We attempted to account for missing data by replacing it hierarchically. That is, if specific primary data were missing, secondary data were used. Where neither primary nor secondary data were available, such as data on the percent of LCD desktop displays recycled or remanufactured, assumptions were made and a sensitivity analysis was performed. In the cases

2.1 GENERAL METHODOLOGY

where no data were found or reasonable assumptions could not be made, these deficiencies are reported.

Any proprietary information required for the assessment was subject to confidentiality agreements between the Center for Clean Products and Clean Technologies and the participating company. Proprietary data are presented as aggregated data to avoid revealing the source. Further, any averaged process data obtained from fewer than three companies are also aggregated to avoid revealing individual inventory items from individual companies.

2.1.7 Critical Review

Critical review is a technique to verify whether an LCA has met the requirements of the study for methodology, data, and reporting, as defined in the goal definition and scoping phase. A critical review process was maintained in the CDP LCA to help ensure that the following criteria were met:

- the methods used to carry out assessments are consistent with the EPA, SETAC, and ISO assessment guidelines;
- the methods used to carry out assessments are scientifically and technically valid within the LCA framework;
- the data used are appropriate and reasonable in relation to the goals of the study;
- the interpretations reflect the limitations identified and the goals of the study; and
- the study results are transparent and consistent.

A project Core Group and Technical Work Group, both consisting of representatives from industry, academia, and government, and EPA's DfE Work Group provided critical reviews of the assessment. Members of these groups are listed in Appendix C, Critical Review. The Core Group served as the project steering committee and was responsible for approving all major scoping assumptions and decisions. The Technical Work Group and the DfE Work Group provided technical guidance and reviews of all major project deliverables including the final LCA report.

In addition to the critical review process, primary data collected were double-checked with the original source to ensure that their data are presented accurately. Additional details on the data verification process for primary data are presented in Sections 2.3 and 2.5.

2.2 MATERIALS EXTRACTION AND MATERIALS PROCESSING (UPSTREAM LIFE-CYCLE STAGES)

2.2.1 Methodology

The inventories included in the materials extraction and materials processing (upstream) life-cycle stages are those of the major primary materials found in the CRT or LCD monitor, as well as major ancillary materials required to manufacture the monitors. Inventories for electricity generation and fuels, which may be used in several life-cycle stages (e.g., to manufacture the products or to use the products), are also presented.

The inventories for extraction and processing of major materials were obtained from existing LCI databases. Electricity generation inventories were developed for this project from secondary sources that describe the distribution of fuels for different electric grids and the fuel inputs and emission outputs associated with different fuel types. The methodologies for developing these inventories are summarized below.

2.2.1.1 Upstream materials processes

Materials for which upstream processes were included in the CRT and LCD life-cycles were selected based on the decision rules described in Section 2.1.2, as well as the availability of secondary data for those materials. An attempt was made to include materials with a mass greater than or equal to one percent of the overall inputs to the product manufacture, or materials that may contribute to a large amount of energy use or have environmental concern. Tables 2-4 and 2-5 in Section 2.1.2 listed the decision criteria for primary materials/components that are included either as separate processes or as part of another process. The upstream materials processes for which inventory data were obtained for this project are presented in Table 2-7.

To determine which source or sources of secondary data to use, nine LCI databases were evaluated against 11 selection criteria in a technical memorandum presented to the project review teams (see Appendix D). Based on this analysis the project team chose the Environmental Information and Management Explorer (EIME) database and the Database for Environmental Analysis and Management (DEAM), two life-cycle inventory databases developed by the *Ecobilan* (Ecobalance) Group (*Ecobilan*, 1999). EIME was developed by *Ecobilan* specifically for electronics and the electronics industry and covers many of the materials specific to the CDP, while the DEAM database includes materials not covered by EIME. Combined, these databases contain detailed inventories of materials extraction and processing activities for most of the materials of interest in this project.

In the *Ecobilan* inventory for a particular material, the functional unit is a set mass of the material. Inputs and outputs are therefore given in terms of mass or other appropriate unit per unit mass of material. These data were imported into the UT Life-Cycle Design Software Tool discussed in Section 2.1.5, where they were linked to the mass of material used in the manufacture of a display monitor to develop inventories specific to the CDP. The associated amounts of each material were presented in Tables 2-2 and 2-3 in Section 2.1.2. How these materials are linked to other processes in the CRT and LCD profiles are shown in Figures 2-2 and 2-3 at the beginning of this chapter.

The materials inventories from *Ecobilan* also contain data for electricity generation, as appropriate, and in some cases, for transportation. Electricity generation and transportation data that were included in the *Ecobilan* inventories could not be separated from the inventory data for materials extraction and conversion processes. Therefore, when electricity generation and transport data were included in the *Ecobilan* inventories, the electricity and transport data collected specifically for this project were not used with the upstream process data.

Material	CRT	LCD
METALS		
aluminum	✓	1
ferrite	✓	
lead	✓	
nickel alloy (invar)	✓	
steel	✓	✓
POLYMERS		
acrylonitrile-butadiene styrene (ABS)	✓	
high impact polystyrene (HIPS)	✓	
polycarbonate (PC)	✓	✓
polyethylene terphthalate (PET)		1
poly(methyl methacrylate) (PMMA)		1
styrene-butadiene co-polymer	✓	✓
ANCILLARY MATERIALS	•	•
natural gas (used to represent LNG)		1

Table 2-7. Materials having upstream processes included in
the CDP LCA

2.2.1.2 Electric grids

Electricity is used in several processes throughout the life-cycle of the CRT and LCD monitors, and in some instances in large amounts. Therefore, the inventory for electricity generation is included in the scope of this project.

As described in Chapter 1, Section 1.4.2, the geographic boundaries of this project are worldwide for upstream and product manufacturing processes and limited to the United States for the use and end-of-life stages. In addition, most CRT and LCD manufacturing is done in Asia and most product manufacturing data collected in this study were from the United States or Japan, except for two LCD manufacturing data sets collected from Korean manufacturers. Therefore, the inventory associated with electricity generation during manufacturing was based on either the Japanese or U.S. electric grids, depending on the particular process or component being manufactured. Where data were obtained from more than one country for the same process, only one electric grid inventory could be used for a single process. In these cases, the Japanese electric grid was used since the majority of manufacturing data are from Japanese companies. The inventory for electricity generated during use and EOL processing was based on the U.S. electric grid. The methodology and results for the electricity generation inventories are detailed in Appendix E, which presents the electric grid technical memorandum prepared for this project. The inventories were developed by first compiling U.S. inventory data for each of the major generation categories or fuel types, including electricity generated from coal, gas, petroleum, and nuclear fuels. These inventories were then combined with data on net electricity generation by fuel type in the U.S. and Japanese electric grids (Table 2-8) to develop the country-wide electric generation inventories. The inputs and outputs for the electricity generation inventories are presented in terms of mass or radioactivity per kWh of electricity generated. Inventories were not included for hydroelectric and renewable energy generation categories due to the scarcity of data on inputs and outputs for these categories. In addition, renewables account for only a small fraction of total U.S. electricity generation.

Fuel	Net electricity generation			
	United States (percent)	Japan (percent)		
Coal	57	18		
Gas	9	20		
Petroleum	3	21		
Nuclear	20	31		
Hydro	11	9		
Other	<1	1		

Table 2-8. Net electricity generation by fuel type

Sources: U.S.: EIA, 1999a; Japan: EIA, 1997; FEPC, 1996.

Note that the Japanese grid inventory is based on the same fuel-specific inventories developed for the U.S. grid, but uses the average distribution of fuels for the Japanese grid (EIA, 1997, FEPC, 1996). This introduces some uncertainty into the Japanese electric grid since Japanese technologies, efficiencies, and pollution control techniques are likely to differ somewhat from their U.S. counterparts. However, the U.S. fuel-specific inventories were used to conserve project resources, rather than expending considerable effort on collecting inventory data from Japanese utilities.

The electricity generation inventories presented in Appendix E are shown as separate process inventories. However, in the overall analysis, they are linked to the manufacturing, use and EOL life-cycle stages, as appropriate. That is, where electricity is used in a process in any of those life-cycle stages, the inputs and outputs from generating the amount of electricity needed is allocated to that process. Note that the U.S. and Japanese electricity generation inventories developed for this project are not linked to the upstream life-cycle stages for materials used to manufacture CRT and LCD desktop computer displays. Electricity generation data were already included in the upstream material inventories received from *Ecobilan*.

2.2.1.3 Fuels

Several fuels are used in manufacturing and end-of-life processes during the life-cycle of the CRT and LCD monitors, and in some instances in large amounts.¹ Therefore, fuel production inventories are included in the scope of this project. These inventories are included in the life-cycle stage in which the fuels are actually consumed (e.g., product manufacturing or end-of-life) instead of in the upstream (materials processing) life-cycle stage. The following fuel inventories are included in both the CRT and LCD LCIs:

- natural gas (also used to represent LNG),
- liquified petroleum gas (LPG),
- fuel oil #2 (distillate),
- fuel oil #6 (residual), and
- fuel oil #4 (average of residual and distillate).

Fuel inventories were obtained from *Ecobilan*. In the *Ecobilan* inventories, the functional unit is a set mass of the material, with inputs and outputs given in terms of mass or other appropriate unit per mass of material (product) produced. These data were imported into the UT Life-Cycle Design software tool where they were linked to the mass of fuel used in different processes in the life-cycle of a display. How the fuel processes are linked to other processes in the CRT and LCD profiles was shown in Figures 2-2 and 2-3.

2.2.2 Data Sources and Data Quality

2.2.2.1 Upstream material and fuel processes

Table 2-9 summarizes data source and data quality information for the data received from *Ecobilan*, which are all secondary data for the purposes of the CDP. In addition to information about CDP data quality indicators (e.g., original source of data, year of data, method in which data were obtained, and geographic boundaries), the table lists whether or not electricity generation or transport data were included in the inventories. This information is important because: (1) electricity generation and transportation data are not from the same sources and therefore are not necessarily consistent among data sets; and (2) transportation data are not included in several of the upstream processes, and are thus a data gap for those processes.

As revealed in Table 2-9, the *Ecobilan* data were derived from various sources, including European data sources and U.S. data sources. In addition, the temporal boundaries of the data vary, with some data being as recent as 1998 but others being from as early as 1975. Electricity generation data are included in all of the inventories, but transportation data are only included in six of 16 data sets. All of these factors create some inconsistencies among the data sets and reduce the data quality when used for the purposes of the CDP. However, this is a common difficulty with LCA, which often uses data from secondary sources for upstream processes to avoid the tremendous amount of time and resources required to collect all the needed data.

¹ Fuels are also used in the materials processing life-cycle stage, but fuel production processes should already be accounted for in the materials inventories obtained from *Ecobilan*.

Material	Electricity	Transport	Year of	Original source ^b
	generation included?	included?	data	
METALS				
aluminum	Y			ETH
ferrite	Y			not provided
lead	Y		Unknown	ETH
nickel-alloy (invar) ^c	Y	Y (nickel)	1991 (nickel)	ETH
steel	Y		1975-1990 ^d	FOEFL, others ^d
POLYMERS	-		_	
acrylonitrile-butadiene styrene (ABS)	Y	Y ^e	1997	Boustead
high impact polystyrene (HIPS)	Y	Y ^e	1997	Boustead
polycarbonate (PC)	Y	Y ^e	1997	Boustead
polyethylene terphthalate (PET)	Y	Y ^e	1998	Boustead
poly(methyl methacrylate) (PMMA)	Y	Y ^e	1997	Boustead
styrene-butadiene co-polymer ^f	Y	Y ^e	1997	Boustead
FUELS				
natural gas	Y	Y	1987-98	six sources cited
liquified petroleum gas (LPG)	Y	Y	1983-93	seven sources cited
fuel oil #2	Y	Y	1983-93	seven sources cited
fuel oil #6	Y	Y	1983-93	seven sources cited
fuel oil #4 ^g	Y	Y	1983-93	seven sources cited

Table 2-9.	Data sources an	d data qu	uality for	the Ecobilar	<i>n</i> inventories ^a
------------	-----------------	-----------	------------	--------------	-----------------------------------

Y: yes, included in inventory.

--: not included in inventory.

^a In general, the *Ecobilan* inventories provide descriptions of data quality but often do not report how data were collected (e.g., measured, estimated, etc.). Therefore, information on the data collection method is not presented here.

^b Sources: ETH (EidgenÖssische Technische Hochschule): Data from the ETH (Swiss Federal Institute of Technology) (*Ecobilan*, 1999). FOEFL (Swiss Federal Office of Environment, Forests and Landscape): Data from the Eco-inventory of Packaging published by the Swiss FOEFL; FOEFL is also known as BUWAL, the acronym in German (*Ecobilan*, 1999). Boustead: LCI database developed by Boustead Consulting (*Ecobilan*, 1999).

In general, the geographic boundaries for different sources are as follows: (1) ETH and FOEFL data are from Europe; (2) Boustead data are from Europe and/or the United States; and (3) miscellaneous sources may be European or U.S. data.

^c The invar inventory is a combination of 36% of the nickel inventory and 64% of the ferrite inventory.

^d The steel inventory was originally provided by *Ecobilan* without detailed documentation; however, DEAM data that were received later have inventories for several steel production processes. The sources listed here are for the DEAM data.

^e Boustead addresses transportation; however, the extent to which it is included in a particular process inventory is uncertain.

^f The styrene-butadiene process is the 50/50 average of the styrene and butadiene processes. ^s The field $\frac{1}{2}$ the 50/50 average of the styrene and butadiene processes.

 $^{\rm g}\,$ The fuel oil #4 process is the 50/50 average of the fuel oil #2 and fuel oil #6 processes.

2.2.2.2 Electric grids

Several sources of data were consulted to generate the fuel-specific inventories that were used to create the overall electricity generation inventories. Table 2-10 summarizes the data sources and some of the data quality indicators for these inventories. Appendix E discusses the data sources and data quality in detail.

As shown in Table 2-10, the electricity generation data were obtained primarily from secondary sources and include data from the mid-1990s as well as data from an unknown time frame. Most are based on measured data collected by the original source, although some are estimated or the data collection method is unknown. Finally, most of the fuel-specific inventories are based on U.S. data, indicating these data are probably less representative and thus of lower quality when applied to the Japanese electric grid.

Type of data	Source	Year of publication	Data collection method	Geographic boundaries
Net electricity generation by fuel	U.S. Energy Information Administration (EIA)	1997	Measured	U.S. and Japan
Primary inputs (Fuel)	EIA	1997	Measured	U.S.
Ancillary inputs	EIA, California Energy Commission, utility contacts	Varies	Varies ^a	U.S.
Air emissions	Primarily AP-42 plus other sources ^b	Varies, but mostly AP-42 data from 1995	Varies ^a	U.S.
Water releases	Oak Ridge National Laboratory (ORNL) °	1994	Unknown	U.S.
Radioactive air and water releases	ORNL ^d	1995	Measured	U.S.
Solid wastes	ORNL ^e	1994	Unknown	U.S.

Table 2-10. Data sources and data quality indicators for the electric generation inventories

^a Includes emission factors from measured and estimated data, plus data where data collection methods were not reported.

^b AP-42 is the U.S. EPA's compilation of Air Pollutant Emissions Factors (EPA, 1996).

^c Coal-fired water release data from ORNL report addressing the externalities of coal fuel cycles (ORNL, 1994).

^d From ORNL report addressing the externalities of nuclear fuel cycles (ORNL, 1995).

^e Coal-fired solid waste data from ORNL report addressing the externalities of coal fuel cycles (ORNL, 1994); radioactive solid waste data from ORNL report addressing the externalities of nuclear fuel cycles (ORNL, 1995).

2.2.3 Limitations And Uncertainties

The limitations and uncertainties associated with the upstream materials, fuels, and electricity generation inventories are primarily due to the fact that these inventories were derived from secondary sources and thus are not tailored to the specific goals and boundaries of the CDP. Because the data are based on a limited number of facilities and have different geographic and temporal boundaries, they are not necessarily representative of current industry practices or of industry practices in the geographic and temporal boundaries defined for the CDP (see

Section 1.4). These are common limitations and uncertainties of LCA, which strives to evaluate the life-cycle environmental impacts of entire product systems and is therefore limited by resource constraints which do not allow the collection of original, measured data for every unit process within a product life cycle. Despite these limitations and uncertainties inherent in LCA methodology itself, LCA remains useful and, indeed is increasingly used by industry, governments, and other stakeholders as part of a comprehensive decision-making process or to understand broad or general environmental trade-offs.

2.2.3.1 Upstream material and fuel processes

Because they are derived from secondary sources, the upstream materials and fuels inventories used in the CDP do not precisely meet the geographic and temporal boundaries outlined for the CDP. Some data are from Europe, some from the United States, and some a combination of both. The manufacturing data for this project were collected from companies in the United States, Japan, and Korea (see Section 2.3), and the available upstream data may not represent the exact location or type of processing represented in the upstream data inventories. This is a limitation to using secondary data; however, the upstream data are only one portion of the overall inventory of the product systems being evaluated, and the project partners chose to focus on collecting primary data for the product manufacturing life-cycle stage, since those data had not been previously compiled.

Another limitation of the upstream inventory data is the lack of transportation data for some processes. These data become particularly important when, for example, raw materials are uncommon and must be transported long distances for processing or when the particular transport mode used for a particular material tends to have high environmental impacts. However, the original data sources used in the *Ecobilan* inventories (see Table 2-9) are among the most used LCI databases in the world (*Ecobilan*, 1999), which suggests the lack of transportation data for upstream processes is not unique to the CDP LCI, but a common limitation of other LCIs as well.

2.2.3.2 Electricity generation data

The limitations to the electricity generation inventory data are provided in Appendix E, Section 6. As another limitation, the Japanese grid was chosen when manufacturing data were from more than one country. Appendix E also describes how the U.S. fuel-specific inventories are applied to the Japanese grid, although technologies, efficiencies, etc. used in Japan are likely to differ from those in the United States. Furthermore, U.S. fuel-specific inventories were derived from secondary sources which did not necessarily meet our temporal boundaries, but did meet geographic boundaries for the U.S. inventory.

2.3 PRODUCT MANUFACTURING

2.3.1 Methodology

2.3.1.1 Identification of processes and manufacturers

Through literature research and contacts with industry experts, the manufacturing processes and the component parts of a CRT and an LCD computer monitor were identified. The major manufacturing processes and components, in terms of resources used, and potential importance to environmental impacts, were selected for inclusion in our primary data collection effort.

Once the components and processes were chosen, companies who might supply manufacturing data for the project needed to be identified. In order to identify those manufacturers, the DfE project's Core and Technical Work Groups were consulted. These groups consist of parties interested in the project results and willing to provide technical assistance throughout the project, including identifying contacts in manufacturing facilities. Manufacturing of CRTs, LCDs, and their component parts is done all over the world. Some manufacturers are in the United States; however, most manufacturers of desktop computer monitors and their components are in Asia. Where available, U.S. industry partners provided contacts at U.S. as well as some Japanese manufacturing facilities and questionnaires were sent to those contacts.

To assist in the collection of data in Asia, UT subcontracted with the Asian Technology Information Program (ATIP) to identify company contacts, and to distribute and collect questionnaires from Asian manufacturers. ATIP acted as a liaison between UT and Japanese and Korean companies that participated in the study.

Participation in the study was completely voluntary. Companies were provided with the goals of the study and the potential benefits of their participation. Once a company chose to participate, they were sent data collection questionnaires to complete information about their manufacturing process and to provide their inventory of process inputs and outputs. A copy of the manufacturing data collection questionnaire that was developed for and used in this study is provided in Appendix F.

The manufacturing processes for which primary data were collected are listed below. In parenthesis are the number of individual data sets collected for each process:

CRT monitor:

- CRT monitor assembly (3)
- CRT (tube) manufacturing (3)
- CRT leaded glass manufacturing (3)

LCD monitor:

- LCD monitor assembly (2)
- LCD panel and module manufacturing (7)
- LCD glass manufacturing $(1)^2$
- color filter patterning on front glass (1)
- liquid crystal manufacturing (2)
- polarizer manufacturing (1)
- backlight unit assembly (3)
- backlight light guide manufacturing (1)
- cold cathode fluorescent lamp (CCFL) manufacturing (1)

How these processes are linked to one another in the CRT and LCD life-cycles is presented in Figures 2-2 and 2-3, respectively. The companies that provided data for the foregoing processes are listed in Table 2-11.

Company	Technology	Company	Technology
American Video Glass Company	CRT	Nippon Denyo Co., Ltd.	LCD
T. Chatani and Co., Ltd.	LCD	Nippon Electric Glass Co., Ltd.	CRT
Chisso Corporation	LCD	Polaroid Corporation	LCD
Eizo Nanao Corporation	CRT, LCD	Samsung Electronics	LCD
Harison Electric Co., Ltd.	LCD	Sharp Corporation	LCD
Hoshiden and Philips Display Corporation	LCD	Sony Corporation (Japan)	CRT
Hyundai Electronics Industries Co., Ltd.	LCD	Sony Electronics Inc. (U.S.)	CRT
Iiyama Electric Co., Ltd.	CRT, LCD	Stanley Electric Co., Ltd.	LCD
Matsushita Electric Industrial Co., Ltd.	LCD	Techneglas	CRT
Merck Japan Ltd.	LCD	Toppan Printing Co., Ltd.	LCD
Mitsubishi Electronic Co., Ltd.	LCD	Toshiba Display Technology Co., Ltd.	CRT, LCD

Table 2-11. Companies that provided primary manufacturing datafor the CRT and/or LCD

Another process that was included in the CRT manufacturing stage data was frit manufacturing. We were not able to obtain primary data for this process; therefore, secondary data were collected from EPA documentation and personal contacts (see Appendix G). An inventory for printed wiring boards (PWBs) was also developed and included in the manufacturing stage analysis. The PWB inventory is based on manufacturing of the electronic boards, and does not include the components on the PWBs. The PWB data were obtained from an industry representative who was able to provide general data not necessarily from one facility, but from a combination of facilities, based on his experience (Sharp, 2000). More details about PWB data collection are provided in Appendix G.

² LCD glass manufacturing data were derived from the three sets of CRT leaded glass manufacturing data (modified to remove lead from the inventory).

2.3 PRODUCT MANUFACTURING

The manufacture of some components were not included in the scope of this study because they were either deemed to be of less significance to the overall product inventories, or data could not be obtained. However, all components were included as part of the final assembled monitor even when individual manufacturing inventories were not. For the CRT, the manufacture of the electron gun, deflection yoke, and phosphors were not included as separate processes, and for the LCD, the transistor metals/materials, spacers, drivers/driver ICs, and color filters were not included.

2.3.1.2 Data collection questionnaires

Data collection questionnaires were developed by the UT research team and approved by the Technical Work Group to most efficiently collect inventory data needed for the LCA. Appendix F provides a copy of the questionnaire given to product and component manufacturers. The data that were collected include brief process descriptions; primary and ancillary material inputs; utility inputs (e.g., electricity, fuels, water); air, water and waste outputs; product outputs; and associated transportation. Quantities of inputs and outputs provided by companies were converted to mass per unit of product. Transport of materials to and products or wastes from the manufacturing facility were also reported. Details of the transportation analysis for this project are presented in Section 2.6.

A total of 27 product manufacturing questionnaires were collected for 11 different processes. The corresponding countries and the number of data sets from each country are listed in Table 2-12.

Process	Country of origin of data (# of data sets)
CRT monitor assembly	Japan (2), U.S. (1)
CRT (tube) manufacturing	Japan (2), U.S. (1)
CRT leaded glass manufacturing	Japan (1), U.S. (2)
CRT frit manufacturing	generic secondary data from the U.S.
LCD monitor assembly	Japan (2)
LCD panel and module manufacturing	Japan (5), Korea (2)
LCD - glass manufacturing	Japan and U.S. (1)*
LCD - color filter patterning on front glass	Japan (1)
LCD - liquid crystal manufacturing	Japan (2)
LCD - polarizer manufacturing	Japan (1)
LCD - backlight unit assembly	Japan (3)
LCD - backlight light guide manufacturing	Japan (1)
LCD - cold cathode fluorescent lamp (CCFL) manufacturing	Japan (1)
PWB manufacturing (for CRT and LCD monitors)	generic secondary data from the U.S.

 Table 2-12. Location of companies and number of process data sets

* Average of three data sets for CRT leaded glass manufacturing modified to remove lead from the inventory.

2.3.1.3 Allocation

Data provided by manufacturers may need to be allocated to the products of interest (i.e., 17" CRT or 15" LCD) in three situations:

- data are provided for more than the defined functional unit;
- data are provided on a rate basis instead of per functional unit (product); and
- data are provided for monitors/components of more than one size (i.e., not only 17" CRTs or 15" LCDs).

In some cases, allocation was not required, as the inventory data collected were for one unit of the product, defined as the functional unit. The three cases where allocation was necessary are briefly described below.

In the first case, simple scaling was required when data were provided for all 17" CRT or 15" LCD monitors produced at a plant, as opposed to only one monitor. This simply requires dividing the inventory mass by the number of monitors produced.

In the second case, data were provided over a certain amount of time. The inventory data were then scaled to represent the functional unit. For example, if it was reported that x kilograms of a material are used *per year* to produce one 17" or 15" monitor, and y number of products are produced per year, then the amount of that material per functional unit is x/y.

In the third situation, allocation was also necessary for a company that manufactured more than just the product or component of interest for this study. For example, a monitor manufacturer may assemble various sized monitors, in addition to 15" LCDs or 17" CRTs. Therefore it was necessary to allocate the process inventory to only our product of interest. We used the difference in mass between the product of interest and other co-products and the difference in the number of each product produced to allocate the inventory to the functional unit.

2.3.1.4 Aggregating manufacturing data

After one set of data from one company is allocated to one monitor, processes for which we collected more than one company's data were averaged together. Once the inventory data for a process were averaged, the electricity consumption from that process was linked to the appropriate electric grid inventory (i.e., Japanese or U.S.). All the manufacturing processes were linked to the Japanese grid, with the exception of frit and PWB manufacturing, both of which were based on data collected in the United States. Where process data were represented by companies in more than one country, the countries in which the majority of facilities were located was used for the basis of which electric grid to use. An exception is the polarizer data, which was from the United States, but the manufacturing was only a pilot plant and not producing a product in the open market. Therefore, it was assumed that polarizer manufacturing, as with most other LCD components, was done in Japan. Once each manufacturing process inventory for each monitor type was complete (i.e., an averaged inventory with an associated electric grid), each was aggregated with the rest of the manufacturing stage processes to comprise the inventory for the manufacturing stage for a monitor. This manufacturing stage inventory was then combined with the other life-cycle stages to represent the full LCI for each monitor type.

2.3.2 Data Sources and Data Quality

While manufacturers worldwide were offered the opportunity to participate in this project, only manufacturing data from Japan, Korea, and the United States were collected. The quality of the data can be evaluated against two factors: (1) the date of the data; and (2) the type of data (i.e., measured, calculated or estimated). An understanding of how data were collected and data verification steps should also be considered when evaluating the data quality. The data collection phase of this project began in 1997 and extended through 2000. Some processes are more sensitive to production dates than others. Most processes included in this analysis are mature technologies and are not expected to differ significantly between the years 1997 and 2000. However, an exception is LCD panel manufacturing, which is an evolving and rapidly advancing process and has seen changes between these years. The countries and dates from which data for each process, most data were from 1998 and 1999.

In the data collection questionnaires, companies identified whether the quantity of each inventory item was a measured, calculated, or estimated value. These identifiers were referred to as the "data quality indicator" (DQI) in the manufacturing questionnaire. The breakdown of DQIs for the inventory items in the CRT and LCD processes are presented in Tables 2-14 and 2-15, respectively. The last line in each table shows overall averages weighted by the number of inventory items in each data set. For the CRT, 43% of the data were measured, 34% calculated, 13% estimated, and 10% were not classified. For the LCD, a similar distribution shows 33% measured, 30% calculated, 23% estimated, and 14% not classified.

Process	# of data sets	Dates of inventory for each data set
CRT monitor assembly	3	1997, 1998-9, 1999
CRT (tube) manufacturing	3	1997, 1998, 1998
CRT leaded glass manufacturing	3	1998, 2000, 2000
LCD monitor assembly	2	1998, 1999
LCD panel and module manufacturing	7	1997-8, 1998, 1998, 1998-9, 1999, 1999, 1999-2000
LCD - color filter patterning on front glass	1	1998
LCD - glass manufacturing*	1	1998-2000
LCD - liquid crystal manufacturing	2	1998, 1998
LCD - polarizer manufacturing	1	1997
LCD - backlight unit assembly	3	1998, 1999, 1999
LCD - backlight light guide manufacturing	1	1999
LCD - cold cathode fluorescent lamp (CCFL) manufacturing	1	1998-9

 Table 2-13. Applicable years of primary data sets

* Primary data, but developed from the CRT glass manufacturing data.

Data quality can also be reviewed in terms of data collection methods. Much effort was given to collecting primary data from manufacturers in this study. Questionnaires were sent out and follow-up communication was conducted to verify data gaps or discrepancies. Twelve companies were visited directly to clarify data and telephone or electronic communication followed-up those and the remaining companies that were providing data for the study. Where data could not be confirmed, additional literature research and discussions with other industry experts were conducted.

Process	% of inventory items that are:			
	Measured	Calculated	Estimated	Not reported
CRT monitor assembly				
Data set 1 (22 inventory items)	9%	64%	27%	0%
Data set 2 (11 inventory items)	9%	83%	0%	9%
Data set 3 (33 inventory items)	3%	0%	0%	<u>97%</u>
total inventory items $= 66$	wt. $avg = 6\%$	wt. $avg = 35\%$	wt. $avg = 9\%$	wt. $avg = 50\%$
CRT (tube) manufacturing				
Data set 1 (69 inventory items)	91%	1%	4%	3%
Data set 2 (51 inventory items)	45%	55%	0%	0%
Data set 3 (83 inventory items)	<u>21%</u>	<u>78%</u>	<u>1%</u>	<u>0%</u>
total inventory items $= 203$	wt. avg = 51%	wt. avg = 46%	wt. avg = 2%	wt. avg = 1%
CRT leaded glass manufacturing				
Data set 1 (43 inventory items)	9%	3%	86%	2%
Data set 2 (45 inventory items)	91%	7%	0%	2%
Data set 3 (2 inventory items)	100%	<u>0%</u>	<u>0%</u>	0%
total inventory items $= 90$	wt. avg = 52%	wt. avg = 5%	wt. avg = 41%	wt. avg = 2%
Overall weighted average for CRT	43%	34%	13%	10%
(359 items)				

 Table 2-14. Data quality indicator percentages for the CRT processes

	% of inventory items that are:				
Process	Measured	Calculated	Estimated	Not reported	
LCD monitor assembly					
Data set 1 (36 inventory items)	5%	78%	17%	0%	
Data set 2 (10 inventory items)	10%	80%	10%	0%	
total inventory items $= 46$	wt. avg = 6%	wt. avg = 78%	wt. avg = 16%	wt. $avg = 0\%$	
LCD panel and module manufacturing					
Data set 1 (71 inventory items)	80%	17%	0%	3%	
Data set 2 (39 inventory items)	0%	100%	0%	0%	
Data set 3 (45 inventory items)	40%	32%	17%	11%	
Data set 4 (139 inventory items)	55%	38%	0%	7%	
Data set 5 (53 inventory items)	43%	0%	36%	21%	
Data set 6 (86 inventory items)	0%	0%	24%	76%	
Data set 7 (32 inventory items)	0%	<u>0%</u>	<u>97%</u>	<u>3%</u>	
total inventory items $= 465$	wt. avg = 37%	wt. avg = 26%	wt. avg = 17%	wt. avg = 20%	
LCD - glass manufacturing*					
(83 inventory items)	0%	0%	100%	0%	
LCD - color filter patterning on front glass					
(29 inventory items)	97%	0%	3%	0%	
LCD - liquid crystal manufacturing					
Data set 1 (41 inventory items)	22%	64%	7%	7%	
Data set 2 (6 inventory items)	<u>0%</u>	<u>0%</u>	<u>0%</u>	<u>100%</u>	
total inventory items $= 47$	wt. avg = 19%	wt. avg = 56%	wt. $avg = 6\%$	wt. avg = 19%	
LCD - polarizer manufacturing					
(30 inventory items)	57%	17%	20%	6%	
LCD - backlight unit assembly					
Data set 1 (20 inventory items)	0%	90%	0%	10%	
Data set 2 (12 inventory items)	50%	0%	8%	42%	
Data set 3 (12 inventory items)	<u>0%</u>	<u>92%</u>	<u>8%</u>	<u>0%</u>	
total inventory items $= 44$	wt. avg = 14%	wt. avg = 66%	wt. $avg = 4\%$	wt. avg = 16%	
LCD - backlight light guide manufacturing					
(5 inventory items)	40%	40%	20%	0%	
LCD - cold cathode fluorescent lamp (CCFL)					
manufacturing (36 inventory items)	58%	39%	0%	3%	
Overall average for LCD (785 items)	33%	30%	23%	14%	

 Table 2-15. Data quality indicator percentages for the LCD processes

* Data based on CRT leaded glass manufacturing data.

2.3.3 Limitations and Uncertainties

The limitations and uncertainties associated with the manufacturing stage are related to the following categories:

- the product system boundaries (scope),
- the data collection process, and
- the data.

Specific limitations/uncertainties for each of these categories are briefly described below.

2.3.3.1 Product system boundary uncertainties

The scope of the analysis included the major monitor components; however, it excluded certain components, such as column and row driver ICs for the LCD and the electron gun for the CRT. The components that were thought to possibly have an effect on the inventory were the column and row drivers, as IC manufacturing is known to be energy intensive and use various process chemicals. Based on some back-of-the-envelope calculations, the exclusion of the manufacturing of the column and row driver ICs is not expected to have a large impact on the inventory or impact results due to the small size of the drivers. Therefore, it is assumed that the exclusion of the column and row drivers will not have a significant impact on the study results.

The scope of the analysis in this study was also dependent on whether companies were willing to provide data. LCD glass manufacturing data from a primary source are not included in the inventory because no companies were willing to supply the data. For example, one manufacturer chose not to provide data because LCD glass manufacturing technology is still developing and is expected to improve from current low yields and high waste generation. However, because glass is an important component by weight of the LCD, we chose to modify the CRT glass manufacturing data to represent LCD glass manufacturing.

Both CRT and LCD glass are considered to be "specialty glasses" in the glass industry and a limited number of companies produce these products. Consequently, there are limited public data available on the production of these glasses. (One major difference between the CRT glass and the LCD glass is that CRT glass contains lead oxide while LCD glass does not.) Therefore, the primary data collected for this study for CRT glass manufacturing was modified by removing inputs and outputs containing lead, and used to represent LCD glass manufacturing. The remaining inputs and outputs were assumed to be the same per kilogram of glass produced. Further research was conducted to confirm whether this was a valid assumption for the energy used in production. Consultation with experts in the field revealed differing estimates between energy used to produce a kilogram of CRT glass and a kilogram of LCD glass. Estimates ranged from an equal amount of energy per kilogram for CRT and LCD glass production, to twice as much energy per kilogram of LCD glass compared to CRT glass. With an assumption of equal or greater energy use for the LCD (call that quantity of energy X), the proportion of that energy (X) that is electrical energy and fuel energy was assumed to be the same as that given in the primary data for CRT glass production (if the portion of energy X for the CRT was 30% electrical and 70% fuel energy, those same proportions were used for the LCD's breakdown of energy X).

2.3 PRODUCT MANUFACTURING

Uncertainty in the differences between energy used for CRT versus LCD glass production are related to production yield and melting point. The melting point of LCD glass is greater than that of CRT glass; however, the difference in the production yield is uncertain compared to the difference in melting point. The production yield for CRT and LCD depends on whether one considers the surface area or the volume of the glass. LCD glass is a flat glass product and excess glass is cut off the ends to obtain the final product. CRT glass manufacturing drops molten glass into a mold and glass is not cut off of a flat piece of glass as in LCD production, but excess glass may be produced during the molding process. Another factor in the difference in energy used for CRT and LCD glass is that the LCD glass must meet high specification standards for use as a substrate for the transistors to be patterned on the glass. Additional finishing steps are required and the process is conducted in clean rooms (described in Section 1.3.3.2, LCD manufacturing). The assumption that the same amount of energy is used for CRT glass and LCD glass production takes into account each of these factors. The baseline analysis in this study assumes the energy use per kilogram of LCD glass is equivalent to that of CRT glass. The uncertainty associated with the glass manufacturing data is a limitation to the manufacturing inventory data set. Additional limitations from the glass manufacturing data are discussed below with other "data uncertainties" in Section 2.3.3.3.

2.3.3.2 Data collection process uncertainties

Limitations and uncertainties related to the data collection process include the fact that companies were self-selected, which could lead to selection bias (e.g., those companies that are more advanced in terms of environmental protection might be more willing to supply data than those that are less progressive). Also, the data were supplied by companies whose vested interest is to have their product look more desirable, which could result in biased data being provided. However, multiple sets of data were obtained for this project, where possible, so that average processes could be developed in an attempt to avoid biased data. The peer review process and employment of the Core and Technical Work Groups as reviewers in this project is intended to help reduce or identify any such bias. Further, several companies were visited and contacted for verification of data.

Other data collection-process limitations resulted from the difficulty in obtaining and verifying data over long distances (i.e., Japan and Korea to United States) as well as from the language barrier. The use of ATIP as the Asian Liaison aided in reducing this limitation; however, there were still language barriers that had to be overcome with ATIP, as well as through direct communication with several companies.

2.3.3.3 Data uncertainties

Additional limitations to the manufacturing stage inventory are related to the data themselves. Several attempts were made to verify or eliminate outliers in the data; however, uncertainty in some data remained due to large data ranges and outliers. Specific data with the greatest uncertainty include: (1) CRT glass manufacturing energy inputs (mentioned above in Section 2.3.3.1); (2) the distribution of fuel/electricity inputs for LCD module manufacturing; and (3) the use of a large amount of liquified natural gas (LNG) as an "ancillary material" and not a fuel.

In addition to the uncertainty in the difference between energy used to manufacture CRT glass and LCD glass, the energy reported to produce a kilogram of CRT glass varied greatly between the data sets. Consultation with experts in the glass industry confirmed that the average energy consumption derived from the primary data sets, although it appeared to be high, could be possible. The total energy inputs per kilogram of glass from the primary data sets used in the analysis, ranged over a factor of approximately 150 (i.e., the largest total energy value in the data was about 150 times that of the smallest value). Due to this large discrepancy, the glass energy data is the subject of sensitivity analyses in this study. The high energy use values were mostly a function of liquified petroleum gas (LPG) used as a fuel.

Other data for which large ranges were found, and which could be important to the results, are the energy use from LCD panel/module manufacturing. Energy data were provided by six of the seven companies supplying LCD panel/module data. The percent of energy from electricity ranged from approximately 3% to 87%. Three of the companies had the electrical energy component contributing greater than 50% of the total energy use, and the other three companies listed other fuels [e.g., LPG, LNG] as contributing greater than 50% to the total energy use. Another large discrepancy was the total energy use for panel/module manufacturing. Four of the six companies had energy use per panel between approximately 440 MJ/panel and 940 MJ/panel, while the two remaining companies had approximately 4,100 and 7,000 MJ/panel. The average per panel was approximately 2,270 MJ/panel and the standard deviation was about 2,910 MJ/panel.

Given the wide variability in the data and large standard deviation, CDP researchers evaluated the data for outliers by breaking the total energy data points into quartile ranges. Minor outliers are then those within a certain range of multipliers beyond the middle 50% of the distribution. That is, the interquartile range (IQR) (i.e., the range of values representing the middle 50%) multiplied by 1.5 is the lower bound of the minor outlier and the IQR multiplied by three is the upper bound. Anything beyond the IQR times three is a major outlier. Using this approach, one data set was found to be a minor outlier and another was found to be a major outlier. These outliers were excluded from the averages used in the baseline analysis, but included in the averages used in a sensitivity analysis (see Section 2.7.3.3).

Finally, the average amount of LNG used as an ancillary material (not a fuel) in LCD panel/module manufacturing was reported as 194 kg per functional unit. This data point remained in the inventory data set for LCD manufacturing, and was assumed to indeed be an ancillary material, and not a fuel. LNG was also reported as a fuel as a separate input (approximately 3.22 kg/functional unit). Keeping the LNG ancillary material in the inventory will not affect the energy impact results, since LNG used as an ancillary material is only linked to the production of that material, and not to the use of it as a fuel. It will, however, affect upstream impacts from the production of the material. Note that the natural gas process was used as a surrogate for the production of LNG, as inventory data was not available for the latter.

2.4 **PRODUCT USE**

2.4.1 Methodology

The methods for developing the use stage inventory are presented in Appendix H and summarized here. CRTs and LCDs use different mechanisms to produce images on screen, which result in different energy use rates. These energy use rates (e.g., kW) can be combined with the time a desktop monitor is on during its lifespan (hours/life) to calculate the total quantity of electrical energy consumed during the use life-cycle stage (e.g., kWh/life). In this project, two lifespan scenarios are considered:

- *Effective life* the actual amount of time a monitor is used, by one or multiple users, before it is disposed of, recycled, or re-manufactured. Reuse of a monitor by a subsequent user is considered part of its effective life. Recycling, on the other hand, is the reuse of parts or materials that require additional processing after disassembly, and it is not considered part of the use stage.
- *Manufactured life* the amount of time either an entire monitor or a single component will last before reaching a point where the equipment no longer functions, independent of user choices.

These two scenarios are considered in this project in order to account for potential differences between how consumers *currently* use the equipment and how consumers could use the equipment. Currently, consumers often replace monitors before they physically break down. This behavior results in a lifespan that is not solely dependent on the monitor technology itself. The manufactured life, on the other hand, is based on the technology and represents how consumers could potentially use the equipment. If the lifespans are significantly different, the difference could have a large impact on how the use stage compares to the other life-cycle stages in this study. The baseline analyses in this project use the effective life scenario and the manufactured life will be part of the sensitivity analyses.

2.4.1.1 Energy use rate

Most desktop monitors manufactured today are built to use several different power consumption modes during normal operation. There are often up to four different power consumption modes that can be used by a monitor in going from a state of active use to a state of almost complete shut-down. These four modes, from greatest power consumption to least, are typically entitled "full-on" or active use, "standby," "suspend," and "active-off." For this report, manufacturers' data on these power modes were collected from company contacts and Internet sites for 35 different 17" CRT monitors and 12 different 15" LCD monitors. The complete list of these data is presented in Appendix H, Attachment A, Table A1.

For the purposes of this study, the power consumption modes have been categorized into two modes: "full-on" and "low." The "low" power mode is an average of the three low power modes typically provided by the manufacturers (i.e., standby, suspend, and active-off). These three categories were averaged to create one "low" power consumption mode because hours per use data (needed for calculations in this study) are only available for a "full-on" and a reduced power mode. The low mode value for the CRT is the average of the three modal averages of standby, suspend, and active-off. For the LCD, data on only two low-power modes (standby and active-off) were provided by manufacturers (see Appendix H, Attachment A, Table A2), and therefore, the low mode value is an average of those two modal averages. Table 2-16 presents the average values for full-on and low power modes that were used for subsequent calculations in this analysis.

Monitor type	Full-on power mode [kilowatts (std. dev.)]	Low power mode ^b [kilowatts (std. dev.)]	
17" CRT	0.113 (0.015 SD)	0.013 (0.005 SD)	
15" LCD	0.040 (0.007 SD)	0.006 (0.003 SD)	

Table 2-16. Average energy use rates ^a

^a See Appendix H, Attachment A, Table A1 for source data.

^b An average of company-reported values for standby, suspend and active-off (see Appendix H, Attachment A, Table A1). Note: 1 kW = 1000 Watts = 1000 J/sec.

2.4.1.2 Effective life (baseline lifespan calculation)

The effective life scenario attempts to model the actual quantity of hours that an average monitor spends in each of the two primary power consumption modes (full-on and a lower power state) during its lifetime. The effective life of an average monitor is based on the following information:

- the proportion of computers that are used in an office environment versus a home environment, to account for different use rates in these two basic user environments;
- the amount of time in a year a typical monitor spends in full-on power mode and in a lower power-consuming mode for both office and home environments; and
- the number of years a typical monitor is used during its lifespan for both office and home environments, not including years in storage before a monitor is replaced or discarded (as it is not consuming power during storage).

Under the effective life scenario, we assume there is no difference in the amount of time a CRT or LCD monitor is operating. That is, the hours per life for the effective life calculation is not technology-dependent. Therefore, the same set of hours-per-life values are used to calculate the kWhs used per effective lifetime for a CRT and an LCD. The remainder of this section discusses the data and methods used to calculate the hours-per-life values used in the effective life scenario. More details are also provided in Appendix H.

Percentages of Office- and Home-Environment Users

Home and office users of computer equipment do not follow the same use patterns. Thus, data are needed on the percent of users in each environment to determine the use pattern of an "average" computer monitor. It is assumed that 65% of computers are in office environments and 35% in home environments, based on data available through the Computer Industry Almanac (CIA, 1997) and the Energy Information Administration (EIA, 1999b) (see Appendix H, Section 2.2.2.1 for more details).

2.4 PRODUCT USE

Note that an "office" environment may be a school, hospital, or other commercial environment, and the computers they use may follow widely varying degrees of use. For example, computers (and thus monitors) in a school may only be used a few hours in a day, while hospitals might operate theirs nearly constantly. For this study, it is assumed that on average, typical office use patterns (to be presented below, and in Appendix H, Section 2.2.2.2) are representative of all non-home environment users.

Operating Pattern (average hours in use per year)

In order to determine the amount of electricity consumed during a monitor's effective life, we need to know the use operating patterns for both the office and home environments. The "operating pattern" is defined here as the number of hours per year spent in each power mode. The average number of hours per mode per year will be the weighted average of the two user environments (i.e., 65% office, 35% home).

A literature search for computer monitor operating patterns was conducted for both office and home environments and a summary of literature reviewed is presented in Appendix H, Attachment A, Table A3. Based on the literature (Nordman *et al.*, 1996, Fanara, 1999, EIA, 1999b) and other assumptions presented in Appendix H, Section 2.2.2.2, we assume the number of hours per year that office and home monitors are used in each mode are as follows:

• Office:

Full-on power mode:	1,095 hrs/yr
Low power mode:	2,263 hrs/yr
Total:	3,358 hrs/yr

• Home:

Full-on power mode:	522 hrs/hr
Low power mode:	793 hrs/yr
Total:	1,315 hrs/yr

Average Years Per Life

The number of years per life, multiplied by the operating patterns in hours per year (listed above), will result in the hours per effective life. A monitor may be reused in multiple "lives" before reaching its end-of-life. The end-of-life is defined as the point at which the monitor is no longer used for its intended purpose in the physical form in which it was originally manufactured. End-of-life options include indefinite storage (in which case it is not reused after storage), de-manufacturing, recycling, or disposal. A monitor may be stored before being reused; however, this storage time will not affect our use calculations since no electricity is required to operate the monitor during this storage. After its first life as used by the original owner, a monitor might be used by different people and with different PC systems in subsequent lives.

For data on the number of years of use that are in a monitor's lifetime, several sources of information were reviewed (see Appendix H). Based on a recent study by the National Safety Council (NSC, 1999), we assumed that a monitor is used for 4 years in its first life and 2.5 years for its second or subsequent lives. The operating patterns (in hours/year) presented above are

assumed to be the same for all of the 6.5 years of the total effective life. However, in the lives subsequent to the first life, the hours per year values are reduced by the fraction of monitors assumed to be reused. Matthews *et al.* (1997) estimated that 45% of PCs are reused after a first life; thus, the effective life operating pattern values in years of life after the first life are 45% of the values in the first life.

Lifespan estimates from the National Safety Council (NSC, 1999) were specific to CRT monitors; however, they were not specific to LCD desktop monitors. The NSC data did contain estimates of a "Notebook PC," which were two to three years for the first life and one to two years for the remaining lives; however, we expect that desktop LCD monitors will more closely mirror the lifetime estimates of a desktop CRT monitor than that of a notebook PC. Consequently, it was assumed that LCD desktop monitors also spend four years in their first life and 2.5 years in their subsequent lives. Additionally, the NSC document did not attempt to separate those computer systems or monitors that are used in an office versus a home environment. Thus, it was assumed that the same years per life are realized for office and home environments.

Effective Life Estimates (hours per life)

The data presented above are summarized in Table 2-17 and used to estimate the total hours per effective life. The values for hours per year per power mode are assumed to be the operating pattern throughout the first life (first four years). In the remaining lives, the annual operating hours decrease to 45% of the hours in operation during each year in the first life, with the remaining lives lasting a total of 2.5 years. Table 2-17 also presents the total hours per effective life per mode, based on percentage in office and home environments. These values are in bold in Table 2-17 (4,586 and 8,961 hrs per effective life) and will be multiplied by the energy use rates per mode (presented in Table 2-16), to calculate the total energy consumption per effective life for each monitor type.

User environment	Power mode	First life (4 years)		Remaining (2.5 yea	Model totals ^b (hr/effective life)	
		Operating pattern (hr/yr)	Total (hr/4 yrs)	Operating pattern (hr/yr) ^a	Total (hrs/2.5 yrs)	
Office	Full-on	1,095	4,380	493	1,233	5,613
(65%)	Low	2,263	9,052	1,018	2,545	11,597
Home	Full-on	523	2,092	235	588	2,680
(35%)	Low	793	3,172	357	893	4,065
Weighted average ^c	Full-on					4,586
	Low					8,961

 Table 2-17. Effective life values

^a The remaining lives operating pattern is 45% of first life operating pattern, based on 45% of monitors that are reused (Matthews *et al.*, 1997).

^b Modal totals calculated as [(Total for first 4 years) + (Total for remaining 2.5 years)].

^c The weighted averages shown for full-on and low power modes are based on the assumption that 65% of users operate in an office environment and 35% operate in a home environment.

Effective Life Total Energy Consumption (kWh/life)

In order to calculate the total kWhs consumed, first, the energy use rates (kW) were multiplied by the lifespans (hours per life) for each mode and each monitor type. They were then summed for the two power modes to obtain a total kWh/life for each monitor type (Table 2-18).

Monitor type	Power mode	Energy use rate (kW)	EL calculated lifespan (hours/life)	EL energy consumption (kWh/life)
17" CRT	Full-on	0.113	4,586	518
	Low	0.013	8,961	116
	Total		13,547	634
15" LCD	Full-on	0.040	4,586	183
	Low	0.006	8,961	54
	Total		13,547	237

Table 2-18. Effective life electricity consumption

2.4.1.3 Manufactured life (alternative lifespan calculation)

Due to the uncertainty and assumptions associated with the effective life scenario, an alternative scenario is also considered. The manufactured life is defined here as the length of time a monitor is designed to operate effectively for the user. It is the number of hours a monitor would function as manufactured, and is independent of user choices or actions. One way to estimate this manufactured life is to use the mean-time-before-failure (MTBF) specification of a monitor or its components. The CRT MTBF specification dictates the amount of time the display must operate before it reaches its brightness "half-life," or the ability to produce 50% of its initial, maximum brightness. The MTBF value, generally provided in total hours per life of a monitor, is what most final manufacturers or assemblers of personal computer (PC) equipment, including monitor assemblers, typically specify for a component. To meet the specification, suppliers typically calculate the MTBF (a military-based specification) based on component data. Suppliers' test results are usually called the "calculated" MTBF. The MTBF value also depends on which combination of power modes are used during testing, which is referred to as the "duty cycle" and each supplier may use a different duty cycle to test their component.

Additionally, monitor assemblers will often perform their own testing, typically entitled "demonstrated" MTBF. The testing includes sequences where the monitor is "stressed" by quickly switching back and forth from an all black picture to an all white one, or quickly switching individual pixels either on and off or through multiple colors or black and white. Manufacturers typically find that their demonstrated MTBF is on the order of twice as long as the calculated MTBF (McConnaughey, 1999; Douglas, 1999). However, it should be noted that the demonstrated MTBF is not a real-time testing method, as the testing data is used in a complex equation to calculate that "demonstrated" value.

From review of the information obtained on CRT-based monitors (see Appendix H, Attachment A, Table A2), it appears that the CRT (the tube) itself is the limiting component, or the component that 99% of the time determines whether the entire monitor has reached its end-of-life. Thus, from the limited information that was obtained on CRTs, and the limited

confidence that can be instilled in those data, an average of the two ranges obtained on the estimated lifetime of CRTs (10,000 and 15,000 hours) was used as the CRT manufactured lifetime (12,500 hours) (Goldwassar, 1999; Douglas, 1999).

For active matrix LCDs, the components that have the greatest potential to fail first are the display panel itself (including the liquid crystals and thin-film transistors), backlights, driver integrated circuit (IC) tabs, and other smaller components. The backlights and driver IC tabs can be field-replaced, thus their failure does not necessarily represent the end of the monitor's life. However, failure of the liquid crystals or transistors, which would require replacement of the display panel itself, would most likely mean that the monitor cannot be cost-effectively repaired. The MTBFs of all these components appear to have a broad range. For example, different backlight manufacturers reported from as few as 15,000 hours to as many as 50,000 hours (Douglas, 1999; Tsuda, 1999; VP150, 1999). However, it appears that those components that are not field-replaceable (e.g., the LCD panel) have MTBFs in the range of 40,000 to 50,000 hours (Tsuda, 1999; Young, 1999). Thus in this study, the amount of time an LCD monitor would operate during its manufactured life is assumed to be the average of the two non field-replaceable values, or 45,000 hours. In order for a monitor to operate for 45,000 hours, any major field-replaceable parts that have MTBFs less than 45,000 hours will need to be accounted for in this LCA project. For example, assuming the backlights last on average 32,500 hours (the average of the values obtained for backlights), more than one (approximately 1.4, on average) would be needed for every panel during its lifetime. Therefore, in the final CDP LCA, the manufacturing of extra backlights would need to be included in the inventory.

Little information is available on the duty cycles that component manufacturers use to test components. Thus, it is assumed that the average duty cycle used in testing components is 50% of the time tested in full-on mode and 50% in a lower power mode. Table 2-19 shows the values that are used in this study for the hours per manufactured life for the CRT and LCD. Some sensitivity analyses were done and presented in Socolof *et al.* (2000).

Monitor type	Total hours (hours/life)	Mode	Duty cycle (% time spent in each mode during testing)	Hours per mode (hours/life)
17" CRT	12,500	Full-on	50%	6,250
		Low	50%	6,250
15" LCD	45,000	Full-on	50%	22,500
		Low	50%	22,500

 Table 2-19.
 Manufactured life values

To calculate the manufactured life electricity consumption (kWh/life), the energy use rate (kW) is multiplied by the lifespan (hours/life) for each monitor in each power mode (Table 2-20). The LCD manufactured life (45,000 hours) is 3.6 times greater than the CRT manufactured life (12,500 hours). In an LCA, comparisons are made based on functional equivalency. Therefore, if one monitor will operate for a longer period of time than another, impacts should be based on an equivalent use. Therefore, based on equivalent use periods, 3.6 CRTs would need to be manufactured for every LCD. This will be incorporated into the profile analysis for the comparative manufactured life LCA.

Monitor type	Power mode	Energy use rate (kW)	ML calculated lifespan (hours/life)	ML energy consumption (kWh/life)		
17" CRT	Full-on	0.113	6,250	706		
	Low	0.013	6,250	81		
	Total		12,500	787		
15" LCD	Full-on	0.040	22,500	900		
	Low	0.006	22,500	135		
	Total		45,000	1,035		

 Table 2-20. Manufactured life electricity consumption

2.4.1.4 Effective life versus manufactured life

For the CRT monitor, the effective life total hours are 13,547 versus the manufactured life total of 12,500. While this suggests that a CRT can be used longer than is physically possible, it simply reveals our low confidence in these numbers and some of their supporting values, with less confidence in the manufactured life data than the effective life estimates. A more complete discussion of the data quality is presented in Section 2.4.2 and Appendix H.

Assumptions were required several times that could bias these numbers in either direction; however, it is thought that the manufactured life estimate is most likely low based on the other estimates for the overall CRT monitor (see Appendix H, Attachment A, Table A2). However, there was no sound basis for assuming a lower value and thus the above hours per life values were used. It should also be stated that while these numbers are different, they are within an 8% error range of one another, and can be taken to be a near 1:1 ratio, indicating a similar potential lifespan.

For LCDs, the comparison across lifespan scenarios is more consistent with what one would expect, with the manufactured life value of 45,000 hours per life being much greater than the effective life value of 13,547 hours per life. The effective life value reflects the assumption that a user's use habits are not technology-dependent, and would seem to reveal that LCDs are not being used as long as they can physically be (less than a third as long).

The difference between the effective and manufactured lives are important when evaluating all the life-cycle stages for a particular monitor type. If the manufactured life is significantly greater than the effective life, the use stage will have greater impacts, as compared to other life-cycle stages. Therefore, it is important to focus on the lifetime scenario that is most realistic, while still recognizing the potential impacts from another feasible lifespan scenario.

In this project, we will use the effective life as the primary basis for the use stage inventory due to the fact that the effective life data are attempting to obtain a more realistic value for electricity consumed per lifetime, and that we currently have greater confidence in those data versus the manufactured life data. The manufactured life data will be used in one sense as a sensitivity analysis and to discuss potential differences in the use stage impacts based on this alternative lifetime scenario.

2.4.2 Data Sources and Data Quality

Source and quality information for the use stage data are detailed in Appendix H, Table 11. We assigned four categories of data quality ratings: excellent, average, poor, and unknown. In general, data assigned higher quality ratings were directly measured and represent 1998 data. As data required more calculation or estimation, or were found from a previous year, the data quality rating was reduced.

The overall level of use stage data quality is between average and excellent (Appendix H, Table 11). However, a distinct difference can be seen in the average data quality ratings given to manufactured life data (average) and the effective life data (excellent). This implies that greater confidence can be placed in the effective life data than in the manufactured life data. Additionally, the energy use rate data appears to be of average quality.

2.4.3 Limitations and Uncertainties

Details of the limitations and uncertainties associated with the energy use rate, the effective life, and the manufactured life estimates are presented in Appendix H, Section 5.

2.5 END-OF-LIFE

2.5.1 Methodology

A Technical Memorandum, attached as Appendix I, was prepared for this project that provides background on the EOL issues for the CRT and LCD. It also provides details on the methodology used in this project for the EOL life-cycle inventory. This section summarizes the salient points from that memorandum needed to understand the EOL methodology and results. For the EOL analysis, a monitor is assumed to have reached EOL status when:

- it has served its useful life;
- is no longer functional; and/or
- is rendered unusable due to technological obsolescence.

Each of these situations is addressed by either the manufactured or effective lives (defined in Section 2.4).

2.5.1.1 EOL disposition options

The major EOL dispositions considered in this analysis are as follows:

- recycling including disassembly and materials recovery;
- landfilling including hazardous (Subtitle C) and non-hazardous (Subtitle D) landfills;
- remanufacturing including refurbishing or reconditioning (to make usable again); and
- incineration waste to energy incineration.

See Appendix I for further descriptions of these dispositions. Note that reuse is considered part of the use stage and not included as an EOL disposition.

The functional unit in this analysis is one monitor; therefore, the different EOL dispositions were allocated as a probability of one monitor going to a certain EOL disposition. Data were somewhat scarce on the percent of monitors going to each disposition, especially for the LCD monitors. After literature research and consultation with the project's Technical Work Group, as well as various other industry experts, project partners chose best estimates of disposition distributions. Table 2-21 presents the assumptions used for the EOL life-cycle stage dispositions for the CRT and LCD, respectively. An explanation of the assumptions and the sources of the data are presented in Appendix I.

The values in Table 2-21 have been used in the baseline scenarios for the CRT and LCD LCIs. To address the uncertainty in the LCD estimates, a sensitivity analysis was conducted (which is discussed in Section 2.7.3).
Disposition	CRT	LCD
Incineration	15%	15%
Recycling	11%	15%
Remanufacturing	3%	15%
Hazardous waste landfill	46%	5%
Solid waste landfill	25%	50%

 Table 2-21. Distribution of EOL disposition assumptions for the CRT and LCD

Sources: NSC, 1999; EPA, 1998; Vorhees, 2000; CIA, 1997; EIA, 1999b

2.5.1.2 Data collection

Inventory data were needed for each of the EOL dispositions to be included in the lifecycle profiles. Primary data were collected for CRT recycling from three companies. Hazardous/solid waste landfilling and incineration were developed from secondary data obtained from *Ecobilan*. In attempts made to obtain remanufacturing data, it was found that remanufacturing processes span a wide range of activities, from as little as replacing button tops to as extensive as testing and replacing PWBs or transformers. Given the broad range of possibilities, no single set of operations could be identified to adequately represent remanufacturing activities that could be incorporated in our model. Remanufacturing data were, therefore, excluded from the assessment.

Recycling

Companies willing to provide CRT recycling data were given EOL questionnaires, which are similar to the manufacturing questionnaires, but modified as appropriate for the EOL lifecycle stage (see Appendix I). The questionnaires were used as a guide for collecting inventory data. The companies agreed to provide inventory data through personal meetings and telephone conversations rather than completing the detailed questionnaire. As a result, the most critical data were identified by the research team to prioritize data needs, and all the details in the questionnaire may not have been provided.

The three companies contacted were: (1) DMC Recycling; (2) A & B Recycling; and (3) The Oak Ridge National Recycle Center (TORNRC). DMC shreds the complete monitor up and separates the recovered materials into three major material streams: ferrous, silica-based, and copper-based. The ferrous metals are sent to steel mills for recycling, while the other streams are sent to lead and copper smelters, respectively. A & B Recycling performs a partial disassembly of the casing and other materials outside the CRT. These materials (namely, HIPS, steel, aluminum, and copper wiring) are sent for recycling, while the CRT itself is shipped to Envirocycle (a CRT recycler in Pennsylvania), where the glass is recovered and sent for recycling back into CRT glass, and the other materials are also recovered for recycling. TORNRC conducts complete monitor disassembly (which includes the CRT recycling process similar to the one performed at Envirocycle), and recovers the individual materials for subsequent recycling.

None of the recycling companies contacted have yet encountered end-of-life LCDs in any appreciable quantities that would justify the development of a separate recycling process for them. Whatever sporadic quantities of LCDs that they do receive (mainly notebook computer

2.5 END-OF-LIFE

displays) are either sent for refurbishing/resale or are processed along with other electronic equipment, by recovering different materials from them, such as metals, glass, plastics, etc. In the absence of actual data for LCD recycling, the shredding-and-materials-recovery process followed by DMC Recycling for CRTs was assumed to be suitable for recovering materials from LCDs as well, and was therefore used to model LCD recycling.

Landfilling and Incineration

Generic secondary data were used for the incineration and landfilling processes because when monitors are disposed of, they are combined with multiple waste streams, and data for monitors alone are not readily available. Although data specific to landfilling and incineration operations for monitors alone were not available, DEAM inventories from *Ecobilan* were available for landfilling and incinerating the following major monitor materials (by weight): steel, glass, plastic, and aluminum. These inventories were combined, based on the approximate proportion of each material in a CRT and an LCD, to create individual processes for landfilling and for incineration (for each monitor type). The proportions of these materials in a CRT and an LCD, presented in Table 2-22, are estimates of the final assembled monitor based on the manufacturing inventory data. Note that these proportions are slightly different from the proportion of total inputs per functional unit presented in Section 2.1.2, Tables 2-2 and 2-3 because materials efficiency during production is not accounted for here (in Table 2-22). The majority of the assembled monitors by weight is accounted for in the overall incineration and landfilling processes, as seen in the totals in Table 2-22.

It should be noted that some of the DEAM inventories associated with incineration or landfilling are for generic materials (i.e., glass, plastic), and may not accurately represent the makeup of the material used in the monitors. For example, the glass is not leaded glass, and the plastics may not represent the exact breakdown of plastics in the monitors being modeled in this study (see Section 2.5.3, Limitations and Uncertainties for further discussion).

	1	
Material	CRT	LCD
Glass	43% (9.48 kg)	9% (0.585 kg)
Steel	30% (6.61 kg)	47% (3.055 kg)
Plastic	17% (3.75 kg)	40% (2.60 kg)
Aluminum	2% (0.441 kg)	1% (0.065 kg)
Total	92% (22.043 kg)	97% (6.5 kg)

Table 2-22. Percent contribution of major materials in the finalproduct

2.5.1.3 Assumptions

The assumptions used in the EOL life-cycle stage include the percentage breakdown of each EOL disposition option (see Table 2-21), the breakdown of materials for the incineration and landfilling inventories (Table 2-22), as well as those listed in Section 2.2 of Appendix I.

2.5.2 Data Sources and Data Quality

Primary CRT recycling data collected were collected from three companies: A & B Recycling, DMC Recycling, and TORNRC. Efforts were made to collect all data in the questionnaires; however, priority was given to obtaining the inventory data. The companies, preferring to provide data over the phone or during personal meetings, were able to provide the inventory data required for the analyses in this study.

Specific DQIs, such as those reported for the manufacturing data (see Section 2.3.2, Tables 2-14 and 2-15), were not obtained. The data from these companies represent facility operations ranging from October 1999 to February 2000. Also, while the data obtained are from three recycling facilities that may have different operations, the averaged inventory data are intended to be representative of various recycling activities in the industry.

Data for the EOL life-cycle stage are a combination of primary data, for which we do not have specific DQI, and secondary data, with limited data quality information. The overall data quality for this life-cycle stage may therefore be limited, and in relative terms, is lower quality than the manufacturing stage data.

Sources of data for EOL distribution assumptions include the National Safety Council (NSC, 1999), EPA (1998), CIA (1997), EIA (1999b) and Vorhees (2000). These are discussed in Appendix I.

2.5.3 Limitations and Uncertainties

Assumptions of the disposition percentages for CRTs and LCDs may not be truly representative of actual dispositions. Recycling technologies are not yet standardized for the sorting, separation, and processing of different types of CRT glass, metals, and plastics. The methods currently employed by a few large-volume recyclers who have been in the CRT recycling business for some years were used in this study and represent "state-of-the-art" in the CRT recycling industry. The LCD recycling process used was based on a CRT recycling process employed by one of the recycling companies contacted, as it was considered general enough to be applicable to LCDs. In the future, when greater numbers of LCDs begin to arrive at recycling facilities, more standardized processes for handling LCDs specifically might be developed.

Limitations for the incineration and landfilling inventories are that incineration and landfilling of the materials were for generic materials and not specific to actually incinerating or landfilling a CRT or LCD monitor. For the CRT, the glass incineration portion of the monitor is for generic, non-leaded glass. The plastics are also generic plastics (mainly those used for packaging that ultimately end up in municipal solid waste, such as HDPE, LDPE, and PET) and may not account for the flame retardants that might be in the plastics, for example. Also, only a few of the major materials by weight are included in the modeled CRT and LCD that are incinerated (listed in Table 2-22).

2.6 TRANSPORTATION

2.6.1 Methodology

Transportation of materials, products, and wastes throughout a product's life-cycle has environmental impacts and should be included in a comprehensive LCA. However, complete transportation data for all life-cycle stages is often difficult to obtain. Only six of the 16 upstream data sets used for this study explicitly stated that transportation was included. An additional six upstream processes are assumed to have considered transportation (Table 2-9). In the manufacturing stage, transportation data were collected in the questionnaires that were distributed to manufacturers. Some transportation data were provided by manufacturers on materials received by their facility and products and wastes shipped from their facility. Data were not obtained on transportation during the use stage because distributing questionnaires and collecting primary data for the use stage were not within the boundaries and scope of the LCA. Consequently, individual consumer transport to pick up purchased monitors and to send to a secondary user or to a recycling/disposal facility were not accounted for. Similarly, EOL data, either from CRT recyclers, or secondary data for incineration and recycling did not include transportation data. Therefore, the transportation data collected in this study may only represent a small portion of the overall transport in the life of a monitor.

The manufacturing data collection questionnaires (Appendix F) provided space for companies to identify transportation information for each material input, product output, or waste output. The questionnaire asked for the distance traveled, mode of transport (i.e., vehicle type), number of trips per year, and percent capacity of the vehicle containing a particular material of interest. Given this information, the project researchers calculated the total distance traveled for a transportation mode per functional unit.

In order to determine the environmental effects of transport, the total distance traveled must be linked to an inventory associated with a transport mode on a per-distance-traveled basis. *Ecobilan's* DEAM data provided inventories for several vehicles either on a per-distance-traveled basis or a per mass-load, per-distance-traveled basis. Given the maximum load of the vehicle, the latter figures can also be used with the transportation data collected in the manufacturing questionnaires to estimate the total distance traveled per mode per functional unit.

2.6.2 Questionnaire Results

In many cases, companies completing the questionnaires provided partial transportation data. For the CRT processes, manufacturers supplied adequate transportation data for 66% of the materials that are expected to have transportation data. For the LCD processes, 73% of the materials had adequate transportation data to determine the total distance traveled per mode per functional unit. Of the transportation data that were provided, Table 2-23 lists the distribution of transport modes for the CRT and LCD manufacturers. To complete the LCI, transport data from the questionnaires would have to be linked to vehicle inventories, which were available through the DEAM data. However, the vehicle inventories were not linked to the questionnaire data because the questionnaire asked for percent capacity, but did not couple that with the load capacity of the vehicle. As a result, the data were inconsistent and could not accurately be used in the overall product LCI.

Transport mode		CRT	LCD			
	Distribution of modes	Approx. normalized ^a distance traveled per functional unit	Distribution of modes	Approx. normalized* distance traveled per functional unit		
Large diesel truck	61%	3 km	58%	3 km		
Small diesel or gas truck	21%	<1 km	35%	<1 km		
Ocean	16%	37 km	3%	<1 km		
Rail	2%	<1 km	2%	<1 km		
Air	0%	0	2%	52 km		
Total	100%	40 km	100%	56 km		

Table 2-23.	Distribution	of transport	modes and	total	distances	per mode
-------------	--------------	--------------	-----------	-------	-----------	----------

^a Normalized by the percent capacity of a vehicle carrying the material of interest. Note: 1 km = 0.6215 miles.

In order to use the vehicle inventory data, it would have to be assumed that the load capacity assumed by manufacturers when providing the percent capacity was consistent with that in the DEAM vehicle inventory data. However, conducting a review of the DEAM data versus the questionnaire data showed that this was not consistent. The greatest difference (based on relatively crude averages of all the transport data provided) appeared to be for the ocean transport, where the discrepancy was on the order of tens of millions of times different. The other modes appeared to be between 22,000 to 660,000 times. These huge discrepancies puts the linked transportation data into question, making it unreliable for use in this study. When the transportation impacts were run in the analysis, and these factors applied, the transportation impacts appeared to be small compared to the other life-cycle stages, but no real reliable information can be gleaned from these data. Further work is needed in this area to understand the true transportation impacts. For this report, the transportation-related inventories are not included.

Transportation data that can be used from the questionnaires include the modes of transport used and the total distances traveled per functional unit (by mode). These are presented in Table 2-23. The "normalized" distance is the total distance traveled for each material multiplied by the percent capacity of the vehicle that was carrying the particular material of interest. This was done to allocate a portion of the vehicle (and thus a portion of the associated inputs and outputs for transport in a particular vehicle) to the transport of the material of interest. It should be noted that these percent capacities were assumptions made by the manufacturers who completed the questionnaires and the percent capacity assumptions could have been inconsistently interpreted. Not only are the distances modified to represent only the product of interest, they are scaled to represent one functional unit (i.e., the material, product, or waste associated with one monitor). Thus, the total normalized distance for the CRT is 40 kilometers (km) per functional unit and 56 km per functional unit for the LCD. These numbers are normalized with the intention of linking them to the individual vehicle inventories, but as stated above, this has not been done due to data inconsistencies.

The most frequently used mode of transport for both the CRT and LCD is the large diesel truck, followed by the small gas or diesel truck. However, the largest normalized distance traveled for the CRT is via ocean transport and for the LCD is via air transport. Worth noting again is that these distances were calculated by normalizing the capacity of the vehicle, as assumed by the manufacturer. Although the transport data represent transport of several

2.6 TRANSPORTATION

materials, products, and wastes into and out of manufacturing facilities, the majority of the distances traveled are from the transport of the final assembled monitors. Of all the reported transportation for the manufacturing stage, the distance traveled of a final assembled CRT monitor via the ocean represents 80% of the total distance traveled. For the LCD, 90% of all reported kilometers traveled are for transport of the final assembled LCD monitor via airplane. Transport of the final assembled CRT monitor for all transport modes is 86% of all kilometers traveled per functional unit. Similarly, for the final assembled LCD monitor, approximately 92% of all manufacturing transportation reported is for the final assembled monitor.

2.6.3 Data Sources and Data Quality

Primary data were derived from manufacturing questionnaires and inventory data for transport vehicles were available through DEAM data. However, inconsistencies between the data made it impossible to accurately apply the DEAM inventories to the questionnaire data. Therefore, the data quality is very low and complete transportation inventory data are excluded from the analysis results.

2.6.4 Limitations and Uncertainties

Inconsistencies between data collected in questionnaires and DEAM data made it impossible to use the transportation inventory data as part of the overall life-cycle. From rough estimates based on data received, it is possible that the transportation impacts are not driving overall life-cycle impacts, however, this would need to be investigated further to confirm such a conclusion.

2.7.1 Baseline LCI

This section presents the baseline inventory data for the life-cycles of the CRT and LCD monitors. The baseline scenario meets the following conditions:

- uses the effective life use stage scenario;
- uses the average value of all the energy inputs from the primary data for glass manufacturing;
- assumes LCD glass manufacturing processes use the same amounts of energy as CRT glass manufacturing per kilogram of glass produced;
- excludes two outliers from the average of the energy inputs in the LCD panel/module manufacturing inventory;
- excludes transportation in the manufacturing stage, but includes any transportation embedded in upstream data sets; and,
- includes the manufacturing process of materials used as fuels (e.g., natural gas, fuel oils) in the life-cycle stage in which they are consumed instead of in the materials processing stage. In cases where materials normally considered to be fuels are used as ancillary materials their manufacturing processes are included with other upstream processes.

Inventory data presented here are used to calculate impacts in the impact assessment (Chapter 3), which translates inventory items into impacts. Note that only limited conclusions can be made based on the inventory alone.

Table 2-24 presents the total quantity of inputs and outputs for the entire life-cycles of the CRT and LCD based on input and output types. Definitions of the input and output types were presented in Table 2-1 in Section 2.1.1. Graphs depicting selected input and output types, derived from the values in Table 2-24, are in Figures 2-5, 2-6, and 2-7. Complete inventory tables for each input and output type by life-cycle stage for the CRT and LCD are provided in Appendix J. The inventories presented in Appendix J list each individual input or output alphabetically for a particular input or output type. The individual inputs or outputs may be the sum of that material for several processes.

The total inventory results for life-cycle inputs reveal that more primary materials,³ water, fuels, electricity, and total energy (i.e., fuel energy plus electricity) are used throughout the CRT life-cycle while more ancillary materials are used throughout the LCD life-cycle. For the life-cycle outputs, the CRT releases more air emissions; water pollutants; hazardous, solid, and radioactive waste; and radioactivity than the LCD. The LCD releases more total wastewater than the CRT. The data that comprise the inventory totals presented in Table 2-24 are listed in Appendix J and broken down by life-cycle stage. Further details on the inventory are provided for each monitor type below.

³ Note that the total mass of primary materials includes the inputs to each process, which may duplicate materials used in processes subsequent to other processes. For example, the primary materials used in steel production are added to the steel used as a primary material for monitor assembly.

Interaction	LCD	TI:4-	
Inputs	CRI	LCD	Units
Primary materials	6.53e+02	3.63e+02	kg/functional unit
Ancillary materials	1.98e+01	2.08e+02	kg/functional unit
Water	1.31e+04	2.82e+03	kg (or L)/functional unit
Fuels	4.33e+02	3.86e+01	kg/functional unit
Electricity	2.49e+03	1.20e+03	MJ*/functional unit
Total energy	2.08e+04	2.84e+03	MJ*/functional unit
Outputs			
Air pollutants	6.64e+02	3.46e+02	kg/functional unit
Wastewater	1.52E+03	3.13e+03	kg (or L)/functional unit
Water pollutants	2.09E+01	1.68e+00	kg/functional unit
Hazardous waste	9.46e+00	6.29e+00	kg/functional unit
Solid waste	1.72e+02	5.23e+01	kg/functional unit
Radioactive waste	2.90e-03	1.48e-03	kg/functional unit
Radioactivity	8.98e+07	4.01e+07	Bq/functional unit

Table 2-24. Total life-cycle inventory summary - baseline analysis

* 3.6 MJ = 1 kWh

Note: Bold indicates the larger value when comparing the CRT and LCD.







2.7.1.1 CRT inventory results

The total CRT inventory presented in Table 2-24 and Figures 2-5, 2-6, and 2-7 show the inventory from all life-cycle stages combined. The totals by life-cycle stage are presented in Table 2-25, Figures 2-8, 2-9, and 2-10.

Inventory type	Upstream	Mfg	Use	EOL	Total	Units ^a
Inputs						
Primary materials	1.58e+01	4.21e+02	2.19e+02	-3.32e+00	6.53e+02	kg
Ancillary materials	2.11e+00	3.54e+00	3.47e+00	1.07e+01	1.98e+01	kg
Water	5.54e+02	1.14e+04	1.14e+03	-2.73e+01	1.31e+04	kg (or L)
Fuels	??	??	??	??	0.00e+00	kg
Electricity	7.32e+01	1.29e+02	2.29e+03	2.29e-01	2.49e+03	MJ
Total energy	3.66e+02	1.83e+04	2.29e+03	-1.28e+02	2.08e+04	MJ
Outputs						
Air pollutants	3.00e+01	1.83e+02	4.49e+02	2.47e+00	6.64e+02	kg
Wastewater	1.70e+01	1.51e+03	0	-3.65e+00	1.52e+03	kg (or L)
Water pollutants	8.12e-01	2.01e+01	7.02e-02	-6.18e-02	2.09e+01	kg
Hazardous waste	??	??	0	??	9.46e+00	kg
Solid waste	9.55e+00	8.12e+01	8.33e+01	-1.66e+00	1.72e+02	kg
Radioactive waste	4.39e-04	1.80e-04	2.28e-03	2.29e-07	2.90e-03	kg
Radioactivity	3.80e+07	3.78e+06	4.80e+07	4.80e+03	8.98e+07	Bq

 Table 2-25. CRT inventory by life-cycle stage

^a Per functional unit (i.e., one CRT monitor over its effective life).







Considering inputs, Figure 2-8 shows that of the inputs measured in mass, the water inputs in the manufacturing life-cycle stage constitute the majority of the inputs by mass for the entire life cycle. Water inputs from the LPG production process constitute almost 80% of the water inputs for all life-cycle stages. In this inventory, the LPG is used in large quantities as a fuel in CRT glass manufacturing. When considering which life-cycle stage contributes most to an inventory category, the manufacturing stage has the largest inventory by mass for primary materials, ancillary materials, water inputs, and fuel inputs. This is also due to the production of LPG as needed for CRT glass production. Fuel inputs are dominated by the manufacturing stage and electricity inputs are dominated by the use stage. The total energy (which is calculated by converting the mass of the fuel into units of energy and combining the fuel energy with the electrical energy⁴) is dominated by the manufacturing life-cycle stage, again mostly due to the large LPG fuel energy used in CRT glass production (Figure 2-9).

Outputs measured in mass include air emissions, wastewater, water pollutants and hazardous, solid, and radioactive waste. Wastewater, by mass (or volume), constitutes the greatest output; however, wastewater alone will not be used to calculate water-related impacts. Water pollutants are also used to calculate water-related impacts. Of the remaining outputs measured in mass (i.e., air emissions, and hazardous, solid and radioactive waste), which are shown on Figure 2-10, air emissions are the greatest contributor to outputs in mass. Note that radioactivity is measured in Bequerels (Bq) and cannot be compared on the same scale.

Considering each inventory type and their contributions by life-cycle stage, the mass of wastewater and water pollutants are greatest in the manufacturing life-cycle stage (again due to LPG consumption). The outputs of air emissions, hazardous waste, solid waste, radioactive waste, and radioactivity all have the greatest contribution from the use stage.

For the outputs, all the totals represented in Table 2-25 include outputs to all dispositions. For example, water outputs sent offsite to treatment as well as those directly discharged to surface waters are all included. Similarly, hazardous, solid and radioactive waste outputs may be landfilled, treated or recycled. The inventory shows these as totals; however, when impacts are calculated, the dispositions dictate which inventory items will be used to calculate impacts (Chapter 3).

The tables and figures discussed above show the total inventories for particular input or output types by life-cycle stage. Tables in Appendix J list each material that contributes to those totals. Figures 2-11 through 2-23 show the total contribution by life-cycle stage, based on the entire input/output type-specific tables in Appendix J. Summary tables for the CRT (Tables 2-26 through 2-34), developed from the Tables in Appendix J, show the top contributing inventory items to each input or output type. Note that Table 2-28 includes input/output types that are classified together as utilities: water, fuel, electricity and total energy.

⁴ Conversions and calculations of energy impacts are described in the LCIA methodology discussion in Chapter 3.



























CRT Primary Inputs

Beginning with the primary data inputs, Figure 2-11 shows that most primary materials are from the manufacturing and use life-cycle stages. To better understand what some of the top contributing materials are to those life-cycle stage totals, Table 2-26 shows the top 99% of the materials contributing to the total CRT primary input inventory. As shown in Table 2-26, the largest material contributor is petroleum, which is about 54% of all the primary CRT inputs. The petroleum is mostly (>98%) from the LPG production process, which relates back to the LPG needed as a fuel in glass production. The other major contributor to primary material inputs is coal (~27%) which is used to produce electricity consumed in the use stage. More detail on the processes that contribute greatest within the manufacturing stage will be presented below after brief discussions of the life-cycle stage breakdowns for each inventory type. For the complete list of primary materials in the CRT inventory, the total mass, and the mass contribution of each life-cycle stage, see Appendix J-1, Table J-1.

CRT Ancillary Inputs

Observing Figure 2-12, the mass of ancillary CRT inputs in the EOL life-cycle stage was greatest (11 kg/functional unit). The upstream stages had the lowest mass of ancillary inputs compared to the other life-cycle stages. To better understand the materials contributing to those totals, Table 2-27 shows that clay is the greatest contributor by mass at 41% of the total CRT ancillary inputs. Clay is used predominately during EOL incineration and landfilling. See Table J-2 in Appendix J for the complete list of ancillary materials in the CRT inventory.

<u> </u>	<u>/0 UI CIXI</u>		atter lais III	<u>puis (ng/iui</u>	<u>ictional un</u>	10/
Material	Upstream	Mfg	Use	EOL	Total	% of total
Petroleum (in ground)	1.32E+00	3.72E+02	3.80E+00	-1.52E+00	3.75E+02	54.17%
Coal, average (in ground)	3.57E+00	5.15E+00	1.79E+02	1.79E-02	1.88E+02	27.15%
Assembled CRT monitor	0	0	2.20E+01	0	2.20E+01	3.38%
Natural gas	0	1.47E+00	1.40E+01	-8.88E-02	1.54E+01	2.36%
Cathode ray tube (CRT)	0	1.07E+01	0	0	1.07E+01	1.64%
CRT glass, unspecified	0	9.76E+00	0	0	9.76E+00	1.50%
Iron (Fe, ore)	6.90E+00	0	0	0	6.90E+00	1.06%
Steel	2.48E-06	5.16E+00	0	0	5.16E+00	0.79%
Natural gas (in ground)	8.41E-01	3.27E+00	0	-1.64E+00	2.47E+00	0.38%
Sand	0	2.40E+00	0	0	2.40E+00	0.37%
Recycled CRT Glass	0	2.06E+00	0	0	2.06E+00	0.32%
Bauxite (Al2O3, ore)	1.37E+00	0	0	0	1.37E+00	0.21%
Iron scrap	9.46E-01	0	0	0	9.46E-01	0.14%
Polycarbonate resin	0	9.23E-01	0	0	9.23E-01	0.14%
PWB-laminate	0	8.47E-01	0	0	8.47E-01	0.13%
Printed wiring board (PWB)	0	8.47E-01	0	0	8.47E-01	0.13%
Styrene-butadiene copolymers	0	8.27E-01	0	0	8.27E-01	0.13%
РРЕ	0	7.35E-01	0	0	7.35E-01	0.11%

Table 2-26. To	p 99	% of CRT	primary	materials in	puts (kg/fur	<u>ictional un</u>	it)

See Appendix J for complete inventory table.

Material	Upstream	Mfg	Use	EOL	Total	% of total
Clay (in ground)	4.49e-03	0	0	8.19e+00	8.19e+00	41.35%
Sand (in ground)	5.85e-02	2.74e-02	0	2.71e+00	2.80e+00	14.13%
Limestone	0	6.91e-02	2.41e+00	2.41e-04	2.48e+00	12.51%
Limestone (CaCO3, in ground)	8.60e-01	1.08e+00	0	-2.39e-01	1.70e+00	8.58%
Lime	0	3.04e-02	1.06e+00	1.06e-04	1.09e+00	5.49%
Sodium chloride (NaCl, in ground or in						
sea)	7.61e-01	1.26e-02	0	-3.07e-05	7.73e-01	3.90%
Sulfuric acid	0	2.38e-01	0	0	2.38e-01	1.20%
Hydrochloric acid	0	2.36e-01	0	0	2.36e-01	1.19%
Sodium hydroxide	0	1.98e-01	0	0	1.98e-01	1.00%
Pyrite (FeS2, ore)	1.94e-01	0	0	0	1.94e-01	0.98%
Nitric acid	0	1.44e-01	0	0	1.44e-01	0.73%
Ferric chloride	0	1.37e-01	0	0	1.37e-01	0.69%
Calcium Chloride	0	1.27e-01	0	0	1.27e-01	0.64%
Calcium hydroxide	0	9.54e-02	0	0	9.54e-02	0.48%
Hydrofluoric acid	0	8.65e-02	0	0	8.65e-02	0.44%
Hydrogen peroxide	0	8.45e-02	0	0	8.45e-02	0.43%
Ammonium hydroxide	0	7.90e-02	0	0	7.90e-02	0.40%
Pumice	0	7.86e-02	0	0	7.86e-02	0.40%
Ammonium chloride	0	7.76e-02	0	0	7.76e-02	0.39%

Table 2 27	Tan 000/ of CDT a	naillawy matarials in	auts (leg/functional unit)
I able 2-2/.	10p 99% of CR1 a	inclinary materials in	puts (kg/iunctional unit)

Tuble 2 2/1 Top >>/0 of Citer unemaily materials inputs (ng/tubelonal unit)									
Material	Upstream	Mfg	Use	EOL	Total	% of total			
Alkali cleaning agent	0	7.72e-02	0	0	7.72e-02	0.39%			
Iron (Fe, ore)	7.23e-02	0	0	3.41e-03	7.57e-02	0.38%			
Potassium peroxymonosulfate	0	7.06e-02	0	0	7.06e-02	0.36%			
Sulfuric acid, aluminum salt	0	6.75e-02	0	0	6.75e-02	0.34%			
Alkali soda (to neutralize acid waste									
water)	0	5.45e-02	0	0	5.45e-02	0.28%			
Polyethylene glycol	0	5.04e-02	0	0	5.04e-02	0.25%			
Bauxite (Al2O3, ore)	1.10e-03	4.47e-02	0	-1.14e-04	4.57e-02	0.23%			
Nitrogen	0	4.57e-02	0	0	4.57e-02	0.23%			
PWB-solder mask solids	0	4.37e-02	0	0	4.37e-02	0.22%			
Potassium hydroxide	0	4.27e-02	0	0	4.27e-02	0.22%			
Lubricant (unspecified)	4.11e-02	0	0	0	4.11e-02	0.21%			
Chlorine	0	4.03e-02	0	0	4.03e-02	0.20%			
Zinc (Zn, ore)	3.79e-02	0	0	0	3.79e-02	0.19%			
Aluminum Oxide	0	3.37e-02	0	0	3.37e-02	0.17%			
Oil (in ground)	0	0	0	3.35e-02	3.35e-02	0.17%			
Sodium Carbonate	0	3.22e-02	0	0	3.22e-02	0.16%			
Tin (Sn, ore)	2.43e-02	0	0	0	2.43e-02	0.12%			

 Table 2-27. Top 99% of CRT ancillary materials inputs (kg/functional unit)

See Appendix J for complete inventory table.

CRT Utility Inputs

Utility inputs in the CRT life-cycle are presented in the inventory in Table 2-28 and include fuel (kg/functional unit), electricity (MJ/functional unit), and water (kg or L/functional unit) inputs. Figures 2-13, 2-14, and 2-15 show the total fuels, water, and electricity inputs, respectively. The fuel and electricity inputs have also been combined into a total energy input category, shown in Figure 2-16. This is also considered one of the impact categories of the LCIA that will be presented in Chapter 3. Therefore, more details on how it is calculated are available in Chapter 3. Briefly, the mass of the fuels are converted to units of energy and added to the electrical energy quantities (in units of MJ).

Material	Upstream	Mfg	Use	EOL	Total	% of total
Fuels (kg/functional unit):	1	8				
LPG	0	3.51E+02	0	3.03E-03	3.51E+02	81.10%
Natural gas (in ground)	2.76E+00	4.56E+01	0	-2.09E-01	4.82E+01	11.14%
Coal, average (in ground)	2.25E+00	1.36E+01	0	-1.16E-02	1.58E+01	3.66%
Petroleum (in ground)	2.02E+00	9.71E+00	0	-5.77E-02	1.17E+01	2.70%
Fuel oil #6	0	3.68E+00	0	0	3.68E+00	0.85%
Fuel Oil #2	0	1.16E+00	0	0	1.16E+00	0.27%
Natural gas	0	2.44E+00	0	-1.30E+00	1.14E+00	0.26%
Coal, lignite (in ground)	9.73E-01	0	0	0	9.73E-01	0.22%
LNG	0	3.35E-01	0	0	3.35E-01	0.08%
Uranium (U, ore)	1.21E-04	2.29E-04	0	-1.99E-07	3.49E-04	<0.01%
Fuel oil #4	0	1.37E-01	0	-1.38E+00	-1.24E+00	-0.29%
Total fuels	8.00E+00	4.28E+02	0	-2.95E+00	4.33E+02	100.00%
Electricity (MJ/functional unit)):					
Electricity	7.32E+01	1.29E+02	2.29E+03	2.29E-01	2.49E+03	
Water (kg or L/functional unit)):					
Water	5.54E+02	1.14E+04	1.14E+03	-2.73E+01	1.31E+04	
Total energy (fuels and electric	ity, MJ/functio	onal unit):				
Energy	3.66E+02	1.83E+04	2.29E+03	-1.28E+02	2.08E+04	

Table 2-28. CRT utility inputs

Table 2-28 shows that LPG used in the manufacturing stage dominates the fuel inputs. LPG from the manufacturing stage is equal to about 81% of all the fuel inputs in the CRT lifecycle. More detail into the process-specific contributions within the manufacturing stage will be presented below at the end of this section. Electricity inputs, however, are dominated by the use stage (~92% of all electricity throughout the CRT life-cycle). When fuel energy and electrical energy are combined into a total energy input value, the overall energy from manufacturing greatly exceeds that from the use stage (18,300 MJ/functional unit versus 2,290 MJ/functional unit). This is depicted in Figure 2-16.

The other utility listed in Table 2-28 is water. Nearly 87% of the water inputs in the CRT life-cycle are from the manufacturing processes; and nearly 80% are from LPG production alone. The life-cycle stage contributing the next most is the use stage at 8.7%. This is from the water used to generate electricity used during the use stage. The upstream stages only contribute about 4% to the total water inputs for the CRT life-cycle. Table J-3 in Appendix J provides the complete list of inventory items for the CRT.

CRT Air Outputs

Air emissions from the CRT life cycle are dominated from the use stage as seen in Figure 2-17. This indicates that most air emissions by mass are from the generation of electricity used by consumers of the monitors. Nearly 68% of the total life-cycle air emissions by mass (or 450 kg/functional unit) are from the use stage. Carbon dioxide (CO₂) alone constitutes 445

kg/functional unit (or about 66% of all air emissions by mass in the life-cycle and nearly 99% of the use stage air emissions). Table 2-29 reveals the individual contribution of CO_2 and other inventory items that contribute to the top 99.99% of air emissions. (See Appendix J for complete inventory table.) These are organized from the air emissions that are the largest contributors to those that are the smaller contributors. Table J-4 in Appendix J shows the contribution of every air emission in the inventory, organized alphabetically. The next largest air emissions, by lifecycle stage, are emitted during the manufacturing stage, which contribute about 28% to the total life-cycle air emissions. Almost 85% of that is air emissions from the LPG production process. All the air emissions in the CRT inventory are designated as direct emissions to the ambient environment.⁵

Material	Upstream	Mfg	Use	EOL	Total	% of total
Carbon dioxide	2.92e+01	1.79e+02	4.45e+02	2.59e+00	6.55e+02	98.68%
Sulfur dioxide	3.37e-01	1.26e-01	2.49e+00	8.30e-04	2.96e+00	0.45%
Nitrogen oxides	6.99e-03	6.95e-01	1.18e+00	-1.90e-02	1.86e+00	0.28%
Methane	6.40e-02	9.08e-01	6.45e-01	-4.30e-02	1.57e+00	0.24%
Sulfur oxides	5.71e-03	8.20e-01	0	-2.97e-02	7.96e-01	0.12%
Carbon monoxide	4.18e-02	4.58e-01	8.09e-02	-4.17e-03	5.76e-01	0.09%
PM	1.28e-01	1.31e-01	0	-1.88e-02	2.40e-01	0.04%
Nonmethane hydrocarbons, remaining unspeciated	9.97e-02	1.10e-01	0	-1.91e-03	2.08e-01	0.03%
Hydrocarbons, remaining unspeciated	1.28e-02	1.58e-01	0	-6.12e-04	1.70e-01	0.03%
Hydrochloric acid	2.39e-03	1.12e-02	1.08e-01	-1.04e-03	1.20e-01	0.02%
Other organics	5.60e-04	7.83e-02	0	-3.65e-03	7.52e-02	0.01%
PM-10	0	3.15e-03	5.78e-02	4.78e-06	6.09e-02	0.01%
Nitrogen dioxide	5.76e-02	0	0	1.85e-03	5.95e-02	0.01%

 Table 2-29. Top 99.99% of CRT air pollutant emissions (kg/functional unit)

CRT Water Outputs

The volume (or mass) of wastewater released throughout the CRT life-cycle is approximately 1,520 L (kg) per functional unit. Approximately 6% of that is sent to treatment as opposed to direct discharge to surface water (Figure 2-18). The mass of chemical pollutants within the wastewater streams was calculated separately. The total mass of these water pollutants released, presented by life-cycle stage, is shown in Figure 2-19. The manufacturing life-cycle stage contributes the greatest mass of water pollutants with approximately 20 kg per functional unit. This is about 96% of all the water pollutants for the entire life-cycle. The upstream stages have the second greatest mass of water pollutants at nearly 1 kg/functional unit (just under 4%). The use and EOL stages are small contributors, with the EOL being negative due to recovery processes within the EOL stage. Table 2-30 shows the major contributors to the

⁵ Note that some companies may not have reported inventory items associated with all output dispositions, as only some dispositions are used for impact calculations. For example, outputs that are treated or recycled and not directly released to the environment are not used in calculating impacts and may not have been reported. This could be applicable to all output inventories.

water pollutant quantities, and reveals that sodium ion, chloride ions and dissolved solids contribute nearly 85% of all the water pollutants to the entire life-cycle. Greater than 95% of the sodium ion outputs are from LPG production and greater than 78% of the chloride ions are from LPG production. As with other input/output types, LPG production used in glass manufacturing has a large impact on the CRT inventory. For the complete inventory, listing water pollutants alphabetically and subtotaled for each life-cycle stage, see Appendix J, Table J-5. Further details on the manufacturing stage will be provided later.

Tuble 2 50. Top 77.570 of CKT water polititant outputs (Kg/Tubetonal unit)							
Material	Disposition	Upstream	Mfg.	Use	EOL	Total	% of total
Sodium (+1)	surface water	2.90e-01	7.04e+00	0	-3.08e-02	7.30e+00	34.94%
Chloride ions	surface water	4.29e-01	6.48e+00	0	-2.39e-02	6.88e+00	32.95%
Dissolved solids	surface water	5.30e-03	3.62e+00	0	-8.77e-05	3.62e+00	17.36%
COD	surface water	1.00e-02	1.60e+00	0	-3.94e-03	1.61e+00	7.71%
Suspended solids	surface water	7.72e-03	8.69e-01	0	-2.11e-03	8.74e-01	4.19%
BOD	surface water	3.93e-04	1.95e-01	0	-4.65e-04	1.95e-01	0.93%
Waste oil	surface water	3.65e-03	1.01e-01	0	-3.13e-04	1.04e-01	0.50%
Dissolved solids	treatment	0	8.01e-02	0	0	8.01e-02	0.38%
Sulfate ion (-4)	treatment	0	1.09e-03	6.84e-02	6.84e-06	6.95e-02	0.33%
Sulfate ion (-4)	surface water	3.75e-02	9.61e-04	0	-1.85e-06	3.84e-02	0.18%
Ammonia ions	surface water	3.54e-06	2.76e-02	0	-6.63e-05	2.75e-02	0.13%
Metals, remaining							
unspeciated	surface water	6.74e-04	9.75e-03	0	-3.42e-05	1.04e-02	0.05%
COD	treatment	0	8.33e-03	0	0	8.33e-03	0.04%
Oil & grease	surface water	0	7.46e-03	0	0	7.46e-03	0.04%
Nitrogen	surface water	4.46e-05	7.18e-03	0	0	7.23e-03	0.03%
Calcium (+2)	surface water	4.96e-03	0	0	0	4.96e-03	0.02%
Carbonate ion	surface water	4.83e-03	0	0	0	4.83e-03	0.02%
Phenol	surface water	6.25e-05	3.63e-03	0	-9.41e-06	3.68e-03	0.02%
Fluoride	surface water	1.89e-05	3.45e-03	0	0	3.47e-03	0.02%
Salts (unspecified)	surface water	1.71e-03	1.62e-03	0	-4.21e-06	3.33e-03	0.02%
Suspended solids	treatment	0	1.29e-03	1.78e-03	1.78e-07	3.07e-03	0.01%
Fluorides (F-)	surface water	9.63e-05	2.97e-03	0	-7.72e-06	3.06e-03	0.01%

Table 2-30 .	Top 99.9% of C	CRT water i	ollutant outpu	its (kg/functional	unit)
	10p ///// 01 C	itt mater p	յնումուն նանթա	its (hg/functional	umic)

CRT Hazardous Waste Outputs

The total mass of hazardous waste generated throughout the life-cycle of the CRT (Figure 2-20) is mostly from the amount of the monitor that is assumed to be placed in a hazardous waste landfill (Table 2-31). This 7.2 kg is based on the proportion of monitors assumed to be hazardous waste as determined in Section 2.5 (EOL) and is approximately 87% of the hazardous waste generated in the EOL stage. Compared to the total mass of hazardous wastes produced throughout the CRT life-cycle, the EOL stage contributes about 88%. The disposition of the waste will be used to determine how impacts are calculated in Chapter 3. Figure 2-20 shows what portion of hazardous waste from the upstream and manufacturing stages are negligible, by

comparison. Table 2-31 and Table J-6 in Appendix J list the hazardous wastes and where each hazardous waste is disposed.

Material	Disposition	Upstream	Mfg	Use	EOL	Total	% of total
EOL CRT monitor, landfilled	landfill	0	0	0	7.20E+00	7.20E+00	76.1%
Hazardous waste	landfill	3.85E-04	6.15E-01	0	-1.50E-03	6.14E-01	6.49%
CRT glass, cullet	R/R	0	0	0	4.84E-01	4.84E-01	5.12%
CRT glass, funnel	R/R	0	0	0	2.29E-01	2.29E-01	2.42%
Transformer	R/R	0	0	0	2.28E-01	2.28E-01	2.41%
PWB-waste cupric etchant	R/R	0	2.25E-01	0	0	2.25E-01	2.38%
Printed wiring board (PWB)	R/R	0	0	0	1.46E-01	1.46E-01	1.54%
General hazardous waste	treatment	0	1.24E-01	0	0	1.24E-01	1.31%
PWB-solder dross	R/R	0	6.70E-02	0	0	6.70E-02	0.71%
General hazardous waste	landfill	4.85E-02	0	0	-9.61E-05	4.84E-02	0.51%
PWB-decontaminating debris	treatment	0	1.55E-02	0	0	1.55E-02	0.16%
PWB-route dust	R/R	0	1.20E-02	0	0	1.20E-02	0.13%
PWB-lead contaminated waste	treatment	0	1.16E-02	0	0	1.16E-02	0.12%
oil							
Chrome liquid waste (D007	R/R	0	9.80E-03	0	0	9.80E-03	0.10%
waste)							
Cinders from CRT glass mfg	landfill	0	8.26E-03	0	0	8.26E-03	0.09%
(70% PbO)							
Unspecified sludge	R/R	0	5.56E-03	0	0	5.56E-03	0.06%
Unspecified sludge	landfill	0	5.22E-03	0	0	5.22E-03	0.06%
CRT glass funnel EP dust (Pb)	R/R	0	5.01E-03	0	0	5.01E-03	0.05%
(D008 waste)							
Waste acid (mostly 3% HCl	R/R	0	3.93E-03	0	0	3.93E-03	0.04%
solution)							
Frit	landfill	0	2.99E-03	0	0	2.99E-03	0.03%
Slag and ash	landfill	0	2.47E-03	0	0	2.47E-03	0.03%
Broken CRT glass	landfill	0	1.88E-03	0	0	1.88E-03	0.02%
Hydrofluoric acid	landfill	0	1.78E-03	0	0	1.78E-03	0.02%

 Table 2-31. Top 99.9% of CRT hazardous waste outputs (kg/functional unit)

R/R: recycling/reuse.

See Appendix J for complete inventory table.

CRT Solid Waste Outputs

Figure 2-21 shows that both the manufacturing and use stages contribute significant amounts of solid waste by mass to the CRT life-cycle. The majority of the solid waste is landfilled. In terms of mass, the greatest contributor to the solid waste outputs for the CRT life-cycle is coal waste that is a result of generating electricity (Table 2-32). Therefore, coal waste is predominately in the use stage, which uses the most electricity, but also in the manufacturing stage, and to a much lesser degree in the EOL stage. Note that the electricity generation processes that support the secondary data used were derived from a different source (i.e., *Ecobilan*) and do not include coal waste as an output; however, the equally large amount of solid waste generated from those processes is listed as "slag and ash" in the upstream and manufacturing inventories. Overall, the top 80% of solid waste generated in the CRT life-cycle is from coal waste, slag and ash, dust/sludge, and fly/bottom ash. Note that different inventories

used in this project have varying nomenclature and some of these solid wastes may indeed overlap. Note also that the mass of a CRT monitor that is assumed to be landfilled at the EOL (3.9 kg/functional unit) is only approximately 2% of the total mass of solid waste in the CRT life-cycle. See Appendix J, Table J-7 for the complete CRT solid waste inventory.

Material	Disposition	Upstream	Mfg	Use	EOL	Total	% of
							total
Coal waste	landfill	0	1.46E+00	5.09E+01	5.09E-03	5.23E+01	30.37%
Slag and ash	landfill	9.65E-02	6.66E+01	0	-1.49E+01	5.18E+01	30.06%
Dust/sludge	landfill	0	5.64E-01	1.97E+01	1.97E-03	2.02E+01	11.75%
Fly/bottom ash	landfill	0	3.65E-01	1.27E+01	1.27E-03	1.31E+01	7.59%
Unspecified solid waste	landfill	4.94E+00	0	0	-7.86E-01	4.15E+00	2.41%
EOL CRT Monitor, landfilled	landfill	0	0	0	3.91E+00	3.91E+00	2.27%
Unspecified solid waste	treatment	0	3.66E+00	0	0	3.66E+00	2.12%
Unspecified solid waste	recycle/reuse	3.07E+00	4.33E-01	0	0	3.50E+00	2.03%
Unspecified waste	landfill	0	3.38E+00	0	-1.49E-02	3.36E+00	1.95%
EOL CRT Monitor, incinerated	treatment	0	0	0	3.31E+00	3.31E+00	1.92%
Iron scrap	recycle/reuse	3.43E-01	0	0	2.50E+00	2.85E+00	1.65%
EOL CRT Monitor, recycled	recycle/reuse	0	0	0	2.42E+00	2.42E+00	1.40%
Broken CRT glass	recycle/reuse	0	1.08E+00	0	0	1.08E+00	0.62%
Mixed industrial (waste)	landfill	4.87E-02	1.00E+00	0	-5.12E-04	1.05E+00	0.61%
Slag and ash	recycle/reuse	0	6.85E-01	0	-3.01E-03	6.82E-01	0.40%
EOL CRT Monitor,	recycle/reuse	0	0	0	6.60E-01	6.60E-01	0.38%
remanufactured							
Mining waste	landfill	4.48E-01	0	0	-1.90E-06	4.48E-01	0.26%
Mineral waste	landfill	4.42E-01	2.61E-03	0	-6.76E-06	4.44E-01	0.26%
Carbon Steel Scrap	recycle/reuse	0	0	0	4.10E-01	4.10E-01	0.24%
flame retardant high-impact	recycle/reuse	0	0	0	4.03E-01	4.03E-01	0.23%
polystyrene (HIPS)							
Waste water treatment (WWT)	recycle/reuse	0	3.72E-01	0	0	3.72E-01	0.22%
sludge							
Ferric chloride	recycle/reuse	0	3.69E-01	0	0	3.69E-01	0.21%
CRT glass, faceplate	recycle/reuse	0	0	0	3.54E-01	3.54E-01	0.21%

 Table 2-32. Top 99% of CRT solid waste outputs (kg/functional unit)

CRT Radioactive Waste Outputs

Radioactive waste outputs in the CRT inventory are found only in the electricity generation and cold-rolled steel production process. Therefore, radioactive wastes will be found wherever electricity is used in a process in the CRT life-cycle. Only very small amounts (approximately 0.003 kg/functional unit) of radioactive waste are generated over the entire life-cycle of the CRT (Figure 2-22 and Table 2-33). As expected, the majority of this is linked to the use stage, where most electricity is used in the CRT life-cycle. Low-level radioactive waste (79%) and depleted uranium (20%) are most of the waste, with very small amounts of highly radioactive waste outputs is small, and therefore, Table 2-33 lists all material outputs associated with radioactive waste, in descending order of quantity. Table J-8 in Appendix J lists these in alphabetical order.

Tuble 2 cov etti Tuulouettye wuste surpuis (ig/fuitettohui unit)							
Material	Disposition	Upstream	Mfg	Use	EOL	Total	% of total
Low-level radioactive waste	landfill	4.11E-04	1.38E-04	1.76E-03	1.76E-07	2.31E-03	79.5%
Uranium, depleted	landfill	0	4.15E-05	5.27E-04	5.27E-08	5.69E-04	19.6%
Radioactive waste (unspecified)	landfill	1.88E-05	0	0	0	1.88E-05	0.6%
Highly radioactive waste (Class C)	landfill	8.65E-06	0	0	0	8.65E-06	0.3%
Total radioactive wastes		4.39E-04	1.80E-04	2.28E-03	2.29E-07	2.90E-03	100.0%

 Table 2-33. CRT radioactive waste outputs (kg/functional unit)

CRT Radioactivity Outputs

Radioactivity is also inventoried in this project as isotopes that are released to the environment. Radioactivity is measured in Bequerels and may be released to air, water, or land. The quantity of radioactivity for each life-cycle stage and different dispositions is presented in Figure 2-23. Table 2-34 shows the top 99.9% of the radioactivity outputs.

Material	Disposition	Upstream	Mfg	Use	EOL	Total	% of
							total
Molybdenum-99 (isotope)	treatment	0	3.72e+06	4.73e+07	4.73e+03	5.10e+07	56.75%
Plutonium-241 (isotope)	landfill	3.74e+07	0	0	0	3.74e+07	41.67%
Xenon-133 (isotope)	air	2.43e+03	6.28e+03	3.12e+05	3.12e+01	3.21e+05	0.36%
Tritium-3 (isotope)	treatment	0	2.20e+04	2.80e+05	2.80e+01	3.02e+05	0.34%
Plutonium-240 (isotope)	landfill	1.62e+05	0	0	0	1.62e+05	0.18%
Cesium-135 (isotope)	landfill	1.46e+05	0	0	0	1.46e+05	0.16%
Radon-222 (isotope)	air	1.37e+05	0	0	0	1.37e+05	0.15%
Plutonium-239 (isotope)	landfill	1.14e+05	0	0	0	1.14e+05	0.13%
Xenon-133 (isotope)	treatment	0	3.48e+03	4.43e+04	4.43e+00	4.78e+04	0.05%
Tritium-3 (isotope)	air	3.47e+02	2.95e+03	3.74e+04	3.75e+00	4.07e+04	0.05%
Xenon-133M (isotope)	air	0	1.99e+04	2.07e+04	2.07e+00	4.06e+04	0.05%
Krypton-85 (isotope)	air	1.73e+02	2.08e+03	2.65e+04	2.65e+00	2.87e+04	0.03%

 Table 2-34.
 Top 99.9% of CRT radioactivity outputs (Bq/functional unit)

See Appendix J for complete inventory table.

Radioactivity outputs are related to the generation of electricity and therefore the greatest quantity of radioactivity is from the use stage, as expected. Table J-9 in Appendix J lists the complete inventory.

CRT Manufacturing Stage

The inventory tables that show the specific materials (i.e., those in Appendix J and Tables 2-26 through 2-34) are the sums of the materials from one or more processes within a life-cycle stage. To burrow down deeper into the data, the manufacturing stage inventory data are broken down by process or group of processes. Groups of processes were combined when fewer than three companies provided data for a process or when confidentiality agreements precluded presenting individual process data. The manufacturing process groups are presented in Table

2-35. Also for confidentiality purposes, upstream and EOL data (derived from *Ecobilan's* data) were not broken down by process. Burrowing further into the contributing processes or process groups is necessary for future manufacturing improvement assessments. Burrowing further into process-specific data at the use stage is not necessary because electricity generation is the only process in the use stage.

Process group	Process(es) included
Monitor assembly	monitor assembly
Tube	CRT (tube) manufacturing
Glass/frit	CRT glass manufacturing, frit manufacturing
PWB	PWB manufacturing
Japanese grid	electricity generation - Japanese electric grid
U.S. grid	electricity generation - U.S. electric grid
Fuels	production of fuel oils #2, #4 and #6, LPG, and natural gas

Table 2-35.CRT process groups

Tables 2-36 through 2-44 list the specific inventories for each process group in the manufacturing stage for each input and output type. Figures 2-24 through 2-34 graph the total inventories for each process group for each input and output type. It should be noted that the input/output type that had the greatest contribution in the manufacturing stage compared to other stages was fuel inputs, which also translated into total energy inputs being greatest in the manufacturing stage. Nonetheless, for purposes of showing more detail in the manufacturing stage and allowing for improvement assessments for manufacturers, the individual material contributions for each manufacturing process group are presented below.

Of the total 421 kg of primary materials per functional unit in the manufacturing stage, fuels production contributes the greatest (374 kg/functional unit), followed by monitor assembly (20.1 kg/functional unit), and then tube manufacturing (11.9 kg/functional unit) (Figure 2-24). The specific material contributions are presented in Table 2-36. Only small amounts of ancillary materials are used in the CRT life cycle and the manufacturing stage only contributed a small percentage of the overall ancillary materials in the life-cycle. However, within the manufacturing life-cycle stage, fuels production and PWB manufacturing had the greatest amount of ancillary materials (1.16 kg/functional unit each) (Figure 2-25).

Material	Quantity (kg/ (functional unit)	% of process group total	% of grand total
Process group			
Monitor assembly			
ABS resin	4.24e-01	2.11%	
Aluminum (elemental)	3.60e-01	1.79%	
Audio cable assembly	9.45e-02	0.47%	
Cables/wires	6.12e-02	0.31%	
Cables/wires	3.33e-01	1.66%	
Cathode ray tube (CRT)	1.07e+01	53.30%	
Connector	5.67e-02	0.28%	
CRT magnet assembly	7.56e-02	0.38%	
CRT shield assembly - ASTM A366/CC#2	2.42e-01	1.21%	
Deflection Yoke assembly	1.51e-01	0.75%	
Demagnetic coil - PU coated paper	1.26e-01	0.63%	
Ferrite	1.70e-01	0.85%	
Phosphate ester	8.31e-03	0.04%	
Polycarbonate resin	9.23e-01	4.60%	
Polystyrene (PS, high impact)	1.51e-01	0.75%	
Power cord assembly	1.13e-01	0.57%	
PPE	7.35e-01	3.66%	
Printed wiring board (PWB)	8.47e-01	4.22%	
Solder, unspecified	2.67e-02	0.13%	
Steel	3.45e+00	17.21%	
Styrene-butadiene copolymers	8.27e-01	4.13%	
Tricresyl phosphate	2.30e-02	0.11%	
Triphenyl phosphate	5.29e-02	0.26%	
Video cable assembly	1.13e-01	0.57%	
Total	2.01e+01	100.00%	4.76%
Tube			
Amyl acetate (mixed isomers)	1.20e-03	0.01%	
Aquadag	2.06e-02	0.17%	
Blue Phosphor (ZnS)	3.84e-03	0.03%	
Blue Phosphor (ZnS.Ag.Al)	1.67e-03	0.01%	
CRT glass, unspecified	9.76e+00	81.70%	
Electron gun	1.01e-01	0.84%	
Frit	6.67e-02	0.56%	
Green Phosphor (ZnS)	3.34e-03	0.03%	
Green Phosphor (ZnS.Cu.Al)	1.34e-03	0.01%	
Nickel Alloy (invar)	2.72e-01	2.28%	
Red Phosphor (Y2O2S)	4.65e-03	0.04%	
Red Phosphor (Y2O2S.Eu)	1.33e-03	0.01%	
Steel	1.71e+00	14.30%	
Total	1.19e+01	100.00%	2.84%

 Table 2-36. CRT manufacturing stage primary material inputs

Material	Quantity (kg/ (functional unit)	% of process group total	% of grand total
Process group			
Glass/frit			
Barium Carbonate	2.97e-01	4.52%	
Glass, unspecified	4.91e-02	0.75%	
Lead	4.47e-01	6.82%	
Potassium Carbonate	3.78e-01	5.76%	
Recycled CRT Glass	2.06e+00	31.37%	
Sand	2.40e+00	36.57%	
Sodium Carbonate	4.88e-01	7.43%	
Strontium Carbonate	3.31e-01	5.05%	
Zircon Sand	5.43e-02	0.83%	
Borax	8.00e-03	0.12%	
Lead	4.67e-02	0.71%	
Silica	5.33e-03	0.08%	
Total	6.56e+00	100.00%	1.56%
РШВ			
PWB-laminate	8.47e-01	94.35%	
Solder (63% tin; 37% lead)	5.08e-02	5.66%	
Total	8.98e-01	100.00%	0.21%
Japanese grid			
Coal, average (in ground)	2.28e+00	47.41%	
Natural gas	1.25e+00	25.89%	
Petroleum (in ground)	1.29e+00	26.69%	
Uranium, yellowcake	3.04e-04	0.01%	
Total	4.82e+00	100.00%	1.14%
U.S. grid			
Coal, average (in ground)	2.86e+00	90.97%	
Natural gas	2.23e-01	7.10%	
Petroleum (in ground)	6.07e-02	1.93%	
Uranium, yellowcake	7.74e-05	<0.01%	
Total	3.15e+00	100.01%	0.75%
Fuels			
Petroleum (in ground)	3.70e+02	99.12%	
Natural gas (in ground)	3.27e+00	0.88%	
Total	3.74e+02	100.00%	88.74%
Grand Total	4.21e+02		100.00%

 Table 2-36. CRT manufacturing stage primary material inputs

Material	Quantity (kg/ (functional unit)	% of process group total	% of grand total
Process group	```´	8	
Monitor assembly			
2,2,4-trimethylpentane	1.50e-04	0.72%	
Cyclohexane	1.88e-04	0.90%	
Fluorocarbon resin	3.75e-05	0.18%	
Isopropyl alcohol	1.94e-02	93.40%	
Surfactant, unspecified	1.42e-04	0.68%	
Synthetic resin, unspecified	8.53e-04	4.10%	
Total	2.08e-02	4.79%	0.59%
Tube			
Acetone	3.17e-04	0.04%	
Acrylic Polymer, unspecified	9.13e-03	1.02%	
Alkali cleaning agent	7.72e-02	8.61%	
Alkali soda (to neutralize acid waste water)	5.45e-02	6.08%	
Ammonia	1.19e-04	0.01%	
Ammonium bifluoride	2.04e-03	0.23%	
Ammonium Dichromate	3.50e-05	<0.01%	
Ammonium fluoride	8.91e-04	0.10%	
Ammonium hydroxide	1.41e-03	0.16%	
Ammonium Oxalate	8.92e-05	0.01%	
Ammonium Oxalate Monohydrate	3.16e-04	0.04%	
Boric acid	4.73e-03	0.53%	
Calcium Chloride	1.27e-01	14.18%	
Calcium hydroxide	9.54e-02	10.64%	
Chlorine	4.03e-02	4.50%	
Chromium (VI)	7.63e-05	0.01%	
Dimethyl Formamide	4.36e-05	<0.01%	
Ferric chloride	1.37e-01	15.32%	
HV Carbon (paste)	1.14e-05	<0.01%	
Hydrochloric acid	4.39e-02	4.89%	
Hydrofluoric acid	7.39e-03	0.82%	
Hydrogen peroxide	5.34e-02	5.96%	
Isopentylacetate	1.74e-03	0.19%	
Muratic Acid (drum)	1.87e-03	0.21%	
Nitric acid	8.17e-03	0.91%	
Nitrogen	4.57e-02	5.10%	
Oxalic acid	5.35e-05	0.01%	
Oxygen (Liquid)	7.57e-03	0.84%	
Periodic Acid	2.26e-04	0.03%	
Polyvinyl alcohol	8.11e-03	0.90%	
Polyvinyl Pyrrolidone (PVP)	2.41e-02	2.69%	
Sodium Dichromate	1.05e-04	0.01%	
Sodium Dichromate Dihydrate (VI)	3.10e-05	<0.01%	

Table 2-37. CRT manufacturing stage ancillary material inputs

	sturing stage an	emary material m	Suts
Material	Quantity (kg/ (functional unit)	% of process group total	% of grand total
Process group			
Sodium hydroxide	3.52e-03	0.39%	
Sodium Hypochlorite	9.25e-05	0.01%	
Sodium Metabisulfite	4.67e-03	0.52%	
Sodium Persulfate	3.54e-04	0.04%	
Sulfuric acid	5.58e-02	6.23%	
Sulfuric acid, aluminum salt	6.75e-02	7.53%	
Toluene	4.80e-03	0.54%	
unspecified CRT process material	5.77e-03	0.64%	
Xylene (mixed isomers)	4.80e-04	0.05%	
Total	8.97e-01	100.00%	25.36%
Glass/frit			
Aluminum Oxide	3.37e-02	17.30%	
Cerium Oxide	3.28e-03	1.68%	
Chromium Oxide	5.62e-05	0.03%	
Hydrofluoric acid	7.91e-02	40.61%	
Pumice	7.86e-02	40.37%	
Total	1.95e-01	100.00%	5.51%
PWB			
Ammonium chloride	7.76e-02	6.68%	
Ammonium hydroxide	7.76e-02	6.68%	
Formaldehyde	6.60e-03	0.57%	
Glycol ethers	2.35e-02	2.03%	
Hydrochloric acid	1.92e-01	16.51%	
Hydrogen peroxide	3.10e-02	2.67%	
Nitric acid	1.36e-01	11.70%	
Polyethylene glycol	5.04e-02	4.34%	
Potassium hydroxide	4.27e-02	3.68%	
Potassium permanganate	1.16e-03	0.10%	
Potassium peroxymonosulfate	7.06e-02	6.08%	
PWB-solder mask solids	4.37e-02	3.76%	
Sodium Carbonate	3.22e-02	2.77%	
Sodium hydroxide	1.94e-01	16.71%	
Sulfuric acid	1.83e-01	15.72%	
Total	1.16e+00	100.00%	32.84%
Japanese grid			
Lime	1.35e-02	30.57%	
Limestone	3.06e-02	69.43%	
Total	4.41e-02	100.00%	1.25%
U.S. grid			
Lime	1.69e-02	30.52%	
Limestone	3.85e-02	69.48%	

 Table 2-37. CRT manufacturing stage ancillary material inputs

Material	Quantity (kg/ (functional unit)	% of process group total	% of grand total
Process group			
Total	5.53e-02	100.00%	1.57%
Fuels			
Bauxite (Al2O3, ore)	4.47e-02	3.85%	
Limestone (CaCO3, in ground)	1.08e+00	92.71%	
Sand (in ground)	2.74e-02	2.36%	
Sodium chloride (NaCl, in ground or in sea)	1.26e-02	1.08%	
Total	1.16e+00	100.00%	32.89%
Grand Total	3.54e+00		100.00%

Table 2-37. CRT manufacturing stage ancillary material inputs

Material	Quantity	% of process group total	% of grand total
Process group			
Fuels (kg/functional unit):			
Monitor assembly			
Fuel oil #4	??		ERR
Tube			
Fuel oil #6	3.68e+00	75.86%	
LNG	3.35e-01	6.91%	
Natural gas	8.37e-01	17.23%	
Total	4.86e+00	100.00%	1.14%
Glass/frit			
Fuel oil #2	1.16e+00	0.33%	
Liquified petroleum gas (LPG)	3.51e+02	99.33%	
Natural gas	1.21e+00	0.34%	
Total	3.53e+02	100.00%	82.72%
РШВ			
Natural gas	??		ERR
Fuels			
Coal, average (in ground)	1.36e+01	19.71%	
Natural gas (in ground)	4.56e+01	66.20%	
Petroleum (in ground)	9.71e+00	14.08%	
Uranium (U, ore)	2.29e-04	<0.01%	
Total	6.89e+01	100.01%	16.14%
Grand Total	4.27e+02		100.00%
Electricity (MJ/functional unit):			
Monitor assembly	1.33e+01		10.27%
Tube	3.19e+01		24.68%
Glass/frit	7.40e+01		57.27%
PWB	1.00e+01		7.77%
Total	1.29e+02		100.00%
Water (kg or L/functional unit):			
Monitor assembly	3.51e+01		0.31%
Tube	8.11e+02		7.09%
Glass/frit	0		0.00%
PWB	4.22e+01		0.37%
Japanese grid	4.43e+01		0.39%
U.S. grid	1.82e+01		0.16%
Fuels	1.05e+04		91.69%
Total	1.14e+04		100.00%
Total energy (fuels and electricity, MJ/function	nal unit):		
Monitor assembly	1.90e+01		0.10%
Tube	2.37e+02		1.29%

Table 2-38. CRT manufacturing stage utility inputs

Material	Quantity	% of process group total	% of grand total
Process group			
Glass/frit	1.52e+04		83.23%
РШВ	2.74e+01		0.15%
Fuels	2.79e+03		15.22%
Total	1.83e+04		100.00%

Table 2-38. CRT manufacturing stage utility inputs

Material	Quantity (kg/ (functional unit)	% of process group total	% of grand total
Process group	()		
Тиве			
Carbon monoxide	1.70E-02	50.61%	
Dimethyl Formamide	3.49E-05	0.10%	
Nitrogen oxides	2.17E-03	6.48%	
Nonmethane hydrocarbons, remaining unspeciated	1.59E-04	0.47%	
Sulfur oxides	9.96E-03	29.73%	
Toluene	3.84E-03	11.46%	
Xylene (mixed isomers)	3.84E-04	1.15%	
Total	3.35E-02	100.00%	0.02%
Glass/frit			
Barium	9.33E-10	<0.01%	
Carbon dioxide	2.85E+00	98.45%	
Carbon monoxide	1.64E-04	0.01%	
Chromium	1.39E-07	<0.01%	
Cobalt	1.43E-10	<0.01%	
Copper	6.33E-10	<0.01%	
Fluorides (F-)	2.93E-05	<0.01%	
Lead	3.22E-07	<0.01%	
Manganese	4.67E-10	<0.01%	
Nickel	5.33E-10	<0.01%	
Nitrogen oxides	4.47E-02	1.54%	
PM	1.10E-04	<0.01%	
Sulfur oxides	5.08E-05	<0.01%	
Zinc (elemental)	4.67E-09	<0.01%	
Total	2.90E+00	1.57%	1.59%
РШВ			
Formaldehvde	3.88E-05		0.00%
Japanese grid			
1,1,1-Trichloroethane	6.46E-08	< 0.01%	
1,2-Dichloroethane	4.57E-08	< 0.01%	
2,3,7,8-TCDD	1.68E-14	<0.01%	
2,3,7,8-TCDF	5.84E-14	<0.01%	
2,4-Dinitrotoluene	3.19E-10	<0.01%	
2-Chloroacetophenone	8.00E-09	<0.01%	
2-Methylnaphthalene	2.38E-10	<0.01%	
5-Methyl chrysene	2.51E-11	<0.01%	
Acenaphthene	4.32E-09	<0.01%	
Acenaphthylene	3.29E-10	< 0.01%	
Acetaldehyde	6.52E-07	<0.01%	
Acetophenone	1.71E-08	<0.01%	
Acrolein	3.31E-07	<0.01%	
Anthracene	4.55E-10	<0.01%	

Table 2-39.	CRT	manufacturing	stage air	· emissions
--------------------	-----	---------------	-----------	-------------

Material	Quantity (kg/	% of process group	% of grand total
	(functional unit)	total	
Process group			
Antimony	9.50E-07	<0.01%	
Arsenic	7.01E-07	<0.01%	
Barium	5.18E-07	< 0.01%	
Benzene	1.52E-06	< 0.01%	
Benzo[a]anthracene	8.00E-10	< 0.01%	
Benzo[a]pyrene	4.35E-11	<0.01%	
Benzo[b,j,k]fluoranthene	3.89E-10	<0.01%	
Benzo[g,h,i]perylene	4.30E-10	<0.01%	
Benzyl chloride	8.00E-07	<0.01%	
Beryllium	3.23E-08	<0.01%	
Biphenyl	1.94E-09	<0.01%	
Bromoform	4.46E-08	<0.01%	
Bromomethane	1.83E-07	<0.01%	
Cadmium	1.41E-07	<0.01%	
Carbon dioxide	1.54E+01	99.14%	
Carbon disulfide	1.49E-07	<0.01%	
Carbon monoxide	2.80E-03	0.02%	
Chloride ions	6.14E-05	< 0.01%	
Chlorobenzene	2.51E-08	<0.01%	
Chloroform	6.74E-08	<0.01%	
Chromium (III)	5.52E-07	<0.01%	
Chromium (VI)	1.34E-07	< 0.01%	
Chrysene	5.35E-10	<0.01%	
Cobalt	1.18E-06	<0.01%	
Copper	3.18E-07	<0.01%	
Cumene hydroperoxide	6.06E-09	<0.01%	
Cyanide (-1)	2.86E-06	<0.01%	
Di(2-ethylhexyl)phthalate	8.34E-08	<0.01%	
Dibenzo[a,h]anthracene	2.96E-10	< 0.01%	
Dichloromethane	3.31E-07	<0.01%	
Dimethyl sulfate	5.49E-08	<0.01%	
Dioxins, remaining unspeciated	7.46E-13	<0.01%	
Ethyl Chloride	4.80E-08	< 0.01%	
Ethylbenzene	1.19E-07	< 0.01%	
Ethylene dibromide	1.37E-09	< 0.01%	
Fluoranthene	1.75E-09	< 0.01%	
Fluorene	1.83E-09	<0.01%	
Fluorides (F-)	6.60E-06	<0.01%	
Formaldehyde	1.02E-05	< 0.01%	
Furans, remaining unspeciated	1.19E-12	< 0.01%	

Table 2-39. C	CRT mai	nufacturing	stage air	[•] emissions
---------------	---------	-------------	-----------	------------------------

Material	Quantity (kg/	% of process group	% of grand total
	(functional unit)	total	
Process group			
Hexane	7.66E-08	< 0.01%	
Hydrochloric acid	1.37E-03	< 0.01%	
Hydrofluoric acid	1.71E-04	<0.01%	
Indeno(1,2,3-cd)pyrene	4.48E-10	< 0.01%	
Isophorone	6.63E-07	<0.01%	
Lead (Pb, ore)	4.41E-07	<0.01%	
Magnesium	1.26E-05	<0.01%	
Manganese (Mn, ore)	1.09E-06	<0.01%	
Mercury	1.18E-07	<0.01%	
Methane	8.13E-05	<0.01%	
Methyl chloride	6.06E-07	<0.01%	
Methyl ethyl ketone	4.46E-07	<0.01%	
Methyl hydrazine	1.94E-07	<0.01%	
Methyl methacrylate	2.29E-08	<0.01%	
Methyl tert-butyl ether	3.99E-08	<0.01%	
Molybdenum	1.55E-07	<0.01%	
Naphthalene	2.21E-07	<0.01%	
Nickel	1.47E-05	<0.01%	
Nitrogen oxides	4.07E-02	0.26%	
Nitrous oxide	1.12E-04	<0.01%	
o-xylene	1.93E-08	<0.01%	
Phenanthrene	5.21E-09	<0.01%	
Phenol	1.83E-08	<0.01%	
PM-10	2.00E-03	0.01%	
Propionaldehyde	4.34E-07	<0.01%	
Pyrene	1.26E-09	<0.01%	
Selenium	1.61E-06	<0.01%	
Styrene	2.86E-08	<0.01%	
Sulfur dioxide	8.62E-02	0.56%	
Tetrachloroethylene	4.92E-08	<0.01%	
TOCs, remaining unspeciated	1.97E-04	<0.01%	
Toluene	1.43E-06	<0.01%	
Vanadium	5.71E-06	<0.01%	
Vinyl acetate	8.67E-09	<0.01%	
Xylene (mixed isomers)	4.23E-08	<0.01%	
Zinc (elemental)	5.15E-06	<0.01%	
Total	1.55E+01	100.00%	8.50%
U.S. grid			
1,1,1-Trichloroethane	3.06E-08	<0.01%	
1,2-Dichloroethane	5.73E-08	<0.01%	
2,3,7,8-TCDD	2.05E-14	<0.01%	

 Table 2-39. CRT manufacturing stage air emissions
Material	Quantity (kg/	% of process group	% of grand total
	(functional unit)	total	
Process group			
2,3,7,8-TCDF	7.30E-14	<0.01%	
2,4-Dinitrotoluene	4.01E-10	<0.01%	
2-Chloroacetophenone	1.00E-08	<0.01%	
2-Methylnaphthalene	4.26E-11	<0.01%	
5-Methyl chrysene	3.15E-11	<0.01%	
Acenaphthene	9.06E-10	<0.01%	
Acenaphthylene	3.60E-10	<0.01%	
Acetaldehyde	8.16E-07	<0.01%	
Acetophenone	2.15E-08	< 0.01%	
Acrolein	4.15E-07	<0.01%	
Anthracene	3.11E-10	< 0.01%	
Antimony	6.96E-08	< 0.01%	
Arsenic	5.99E-07	<0.01%	
Barium	3.28E-08	<0.01%	
Benzene	1.86E-06	<0.01%	
Benzo[a]anthracene	1.48E-10	<0.01%	
Benzo[a]pyrene	5.44E-11	<0.01%	
Benzo[b,j,k]fluoranthene	1.70E-10	<0.01%	
Benzo[g,h,i]perylene	5.75E-11	<0.01%	
Benzyl chloride	1.00E-06	<0.01%	
Beryllium	3.05E-08	<0.01%	
Biphenyl	2.43E-09	<0.01%	
Bromoform	5.58E-08	<0.01%	
Bromomethane	2.29E-07	<0.01%	
Cadmium	7.69E-08	<0.01%	
Carbon dioxide	7.10E+00	98.98%	
Carbon disulfide	1.86E-07	<0.01%	
Carbon monoxide	1.29E-03	0.02%	
Chloride ions	2.90E-06	< 0.01%	
Chlorobenzene	3.15E-08	< 0.01%	
Chloroform	8.45E-08	<0.01%	
Chromium (III)	3.88E-07	<0.01%	
Chromium (VI)	1.15E-07	<0.01%	
Chrysene	1.63E-10	<0.01%	
Cobalt	1.94E-07	<0.01%	
Copper	1.59E-08	<0.01%	
Cumene	7.59E-09	<0.01%	
Cyanide (-1)	3.58E-06	<0.01%	
Di(2-ethylhexyl)phthalate	1.04E-07	<0.01%	
Dibenzo[a,h]anthracene	1.40E-11	<0.01%	
Dichloromethane	4.15E-07	< 0.01%	

Material	Quantity (kg/	% of process group	% of grand total
	(functional unit)	total	
Process group			
Dimethyl sulfate	6.87E-08	< 0.01%	
Dioxins, remaining unspeciated	9.33E-13	<0.01%	
Ethyl Chloride	6.01E-08	<0.01%	
Ethylbenzene	1.35E-07	< 0.01%	
Ethylene dibromide	1.72E-09	< 0.01%	
Fluoranthene	1.07E-09	<0.01%	
Fluorene	1.34E-09	<0.01%	
Fluoride	3.12E-07	<0.01%	
Formaldehyde	1.35E-06	<0.01%	
Furans, remaining unspeciated	1.49E-12	<0.01%	
Hexane	9.59E-08	<0.01%	
Hydrochloric acid	1.72E-03	0.02%	
Hydrofluoric acid	2.15E-04	<0.01%	
Indeno(1,2,3-cd)pyrene	1.05E-10	<0.01%	
Isophorone	8.30E-07	<0.01%	
Lead	2.04E-07	<0.01%	
Magnesium	1.57E-05	<0.01%	
Manganese	7.28E-07	<0.01%	
Mercury	1.20E-07	<0.01%	
Methane	1.03E-02	0.14%	
Methyl chloride	7.59E-07	<0.01%	
Methyl ethyl ketone	5.58E-07	<0.01%	
Methyl hydrazine	2.43E-07	<0.01%	
Methyl methacrylate	2.86E-08	<0.01%	
Methyl tert-butyl ether	5.01E-08	<0.01%	
Molybdenum	9.32E-09	<0.01%	
Naphthalene	2.92E-08	<0.01%	
Nickel	1.09E-06	<0.01%	
Nitrogen oxides	1.88E-02	0.26%	
Nitrous oxide	5.42E-05	<0.01%	
o-xylene	9.11E-10	<0.01%	
Phenanthrene	4.00E-09	<0.01%	
Phenol	2.29E-08	< 0.01%	
Phosphorus (yellow or white)	7.91E-08	< 0.01%	
PM-10	9.22E-04	0.01%	
Propionaldehyde	5.44E-07	< 0.01%	
Pyrene	5.32E-10	< 0.01%	
Selenium	1.87E-06	<0.01%	
Styrene	3.58E-08	<0.01%	
Sulfur dioxide	3.98E-02	0.56%	
Tetrachloroethylene	6.16E-08	< 0.01%	

 Table 2-39. CRT manufacturing stage air emissions

Material	Quantity (kg/	% of process group	% of grand total
	(functional unit)	total	
Process group			
TOCs, remaining unspeciated	9.19E-05	< 0.01%	
Toluene	4.06E-07	< 0.01%	
Vanadium	2.81E-07	< 0.01%	
Vinyl acetate	1.09E-08	< 0.01%	
Xylene (mixed isomers)	5.30E-08	< 0.01%	
Zinc (elemental)	2.43E-07	<0.01%	
Total	7.17E+00	100.00%	3.93%
Fuels			
1,1,1-Trichloroethane	1.36E-07	<0.01%	
1,2-Dichloroethane	2.71E-07	<0.01%	
1,4-Dichlorobenzene	3.06E-07	<0.01%	
2,4-Dinitrotoluene	1.90E-09	<0.01%	
2-Chloroacetophenone	4.75E-08	<0.01%	
2-Methylnaphthalene	6.11E-09	<0.01%	
3-Methylcholanthrene	4.59E-10	<0.01%	
5-Methyl chrysene	1.49E-10	<0.01%	
Acenaphthene	5.31E-09	<0.01%	
Acenaphthylene	2.17E-09	<0.01%	
Acetaldehyde	3.86E-06	<0.01%	
Acetophenone	1.02E-07	<0.01%	
Acrolein	1.97E-06	<0.01%	
Aldehydes	1.52E-03	<0.01%	
Aluminum (elemental)	1.98E-05	<0.01%	
Ammonia	2.35E-03	<0.01%	
Anthracene	2.12E-09	<0.01%	
Antimony	6.36E-07	<0.01%	
Aromatic hydrocarbons	5.29E-08	<0.01%	
Arsenic	1.41E-05	<0.01%	
Barium	3.33E-07	<0.01%	
Benzene	1.57E-02	0.01%	
Benzo[a]anthracene	1.27E-09	<0.01%	
Benzo[a]pyrene	6.94E-10	<0.01%	
Benzo[b,j,k]fluoranthene	7.46E-10	<0.01%	
Benzo[b]fluoranthene	5.07E-10	<0.01%	
Benzo[g,h,i]perylene	5.63E-10	<0.01%	
Benzo[k]fluoranthene	5.07E-10	<0.01%	
Benzyl chloride	4.75E-06	<0.01%	
Beryllium	1.42E-06	< 0.01%	
Biphenyl	1.15E-08	< 0.01%	
Bromoform	2.64E-07	< 0.01%	
Bromomethane	1.08E-06	<0.01%	

Material	Ouantity (kg/	% of process group	% of grand total
	(functional unit)	total	
Process group			
Butane	5.35E-04	<0.01%	
Cadmium	8.20E-07	<0.01%	
Calcium	1.72E-05	< 0.01%	
Carbon dioxide	1.54E+02	97.92%	
Carbon disulfide	8.81E-07	< 0.01%	
Carbon monoxide	4.36E-01	0.28%	
Chloride ions	3.39E-05	<0.01%	
Chlorine	5.84E-09	< 0.01%	
Chlorobenzene	1.49E-07	< 0.01%	
Chloroform	4.00E-07	< 0.01%	
Chromium (III)	2.13E-05	< 0.01%	
Chromium (VI)	2.13E-05	< 0.01%	
Chrysene	1.22E-09	< 0.01%	
Cobalt	2.54E-06	< 0.01%	
Copper	1.57E-06	< 0.01%	
Cumene	3.59E-08	< 0.01%	
Cyanide (-1)	1.69E-05	< 0.01%	
Di(2-ethylhexyl)phthalate	4.95E-07	<0.01%	
Dibenzo[a,h]anthracene	3.61E-10	<0.01%	
Dichloromethane	1.97E-06	< 0.01%	
Dimethyl sulfate	3.25E-07	< 0.01%	
Dimethylbenzanthracene	3.82E-09	<0.01%	
Dioxins, remaining unspeciated	1.10E-10	< 0.01%	
Ethane	7.90E-04	< 0.01%	
Ethyl Chloride	2.85E-07	<0.01%	
Ethylbenzene	6.44E-07	<0.01%	
Ethylene dibromide	8.14E-09	< 0.01%	
Fluoranthene	5.74E-09	<0.01%	
Fluorene	7.03E-09	< 0.01%	
Fluorides (F-)	3.84E-06	< 0.01%	
Formaldehyde	1.19E-03	< 0.01%	
Furans, remaining unspeciated	5.11E-10	<0.01%	
Halogenated hydrocarbons (unspecified)	2.94E-13	<0.01%	
HALON-1301	5.11E-10	<0.01%	
Hexane	4.59E-04	< 0.01%	
Hydrocarbons, remaining unspeciated	1.58E-01	0.10%	
Hydrochloric acid	8.14E-03	< 0.01%	
Hydrofluoric acid	1.02E-03	< 0.01%	
Hydrogen sulfide	3.11E-03	<0.01%	
Indeno(1,2,3-cd)pyrene	9.43E-10	<0.01%	
Iron	3.83E-05	< 0.01%	

Material	Quantity (kg/ (functional unit)	% of process group total	% of grand total
Process group			
Isophorone	3.93E-06	<0.01%	
Lead	1.25E-05	<0.01%	
Magnesium	7.46E-05	<0.01%	
Manganese	2.26E-05	< 0.01%	
Mercury	8.81E-07	<0.01%	
Metals, remaining unspeciated	3.16E-07	<0.01%	
Methane	8.98E-01	0.57%	
Methyl chloride	3.59E-06	<0.01%	
Methyl ethyl ketone	2.64E-06	<0.01%	
Methyl hydrazine	1.15E-06	<0.01%	
Methyl methacrylate	1.36E-07	<0.01%	
Methyl tert-butyl ether	2.37E-07	<0.01%	
Molybdenum	1.97E-06	<0.01%	
Naphthalene	3.54E-07	<0.01%	
Nickel	1.24E-04	<0.01%	
Nitrogen oxides	5.88E-01	0.37%	
Nitrous oxide	1.64E-02	0.01%	
Nonmethane hydrocarbons, remaining unspeciated	1.10E-01	0.07%	
n-Propane	1.69E-06	<0.01%	
Other organics	7.83E-02	0.05%	
o-xylene	1.11E-06	<0.01%	
Pentane	6.62E-04	<0.01%	
Phenanthrene	2.30E-08	<0.01%	
Phenol	1.08E-07	<0.01%	
Phosphorus (yellow or white)	1.25E-05	<0.01%	
РМ	1.31E-01	0.08%	
PM-10	2.28E-04	<0.01%	
Polycyclic aromatic hydrocarbons	5.87E-11	<0.01%	
Propionaldehyde	2.58E-06	<0.01%	
Pyrene	3.65E-09	<0.01%	
Selenium	9.47E-06	<0.01%	
Silicon	1.72E-05	<0.01%	
Sodium	1.02E-04	<0.01%	
Styrene	1.69E-07	<0.01%	
Sulfur oxides	8.10E-01	0.52%	
Tetrachloroethylene	2.92E-07	<0.01%	
Toluene	3.81E-06	<0.01%	
Vanadium	2.68E-04	< 0.01%	

	ODT	• • •		•	• •
I able 2-39.	СКТ	manufacturing	stage	air	emissions
	~	manaractaring	suge		emissions

Material	Quantity (kg/	% of process group	% of grand total
	(functional unit)	total	
Process group			
Vinyl acetate	5.15E-08	<0.01%	
Zinc (elemental)	1.02E-05	<0.01%	
Total	1.57E+02	100.00%	85.96%
Grand Total	1.83E+02		100.00%

Material	Disposition	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process group				
WASTEWATER STREAMS				
Tube				
Wastewater stream	treatment	5.26e+01	9.87%	
Wastewater stream	surface water	4.81e+02	90.13%	
Total		5.33e+02	100.00%	35.42%
Glass/frit				
Wastewater stream	surface water	3.62e+01		2.40%
РШВ				
Wastewater stream	treatment	4.22e+01		2.80%
Fuels				
Wastewater stream	surface water	8.94e+02		59.38%
Grand Total		1.51e+03		100.00%
WASTEWATER POLLUTANTS		-		
Tube				
BOD	surface water	6.39e-03	5.34%	
Chromium ore	surface water	1.02e-05	<0.01%	
Chromium ore	treatment	1.03e-06	<0.01%	
COD	surface water	7.22e-03	6.04%	
COD	treatment	8.33e-03	6.97%	
Copper	surface water	1.80e-06	<0.01%	
Cyanide (-1)	surface water	6.06e-07	<0.01%	
Dissolved solids	treatment	8.01e-02	67.03%	
Fluoride	surface water	3.45e-03	2.89%	
Fluoride	treatment	3.51e-04	0.29%	
Iron	surface water	1.65e-04	0.14%	
Lead	surface water	3.01e-06	<0.01%	
Lead	treatment	1.03e-06	<0.01%	
Manganese	surface water	3.60e-06	< 0.01%	
Molybdenum	surface water	1.20e-07	<0.01%	
Nickel	surface water	7.93e-05	0.07%	
Nitrogen	surface water	7.18e-03	6.01%	
Oil & grease	surface water	2.41e-04	0.20%	
Phosphate as P2O5	surface water	1.21e-06	<0.01%	
Phosphorus (yellow or white)	surface water	5.05e-05	0.04%	
Suspended solids	surface water	4.63e-03	3.87%	
Suspended solids	treatment	1.28e-03	1.07%	
Zinc (elemental)	surface water	1.39e-05	0.01%	
Zinc (elemental)	treatment	1.03e-06	<0.01%	
Total		1.20e-01	100.00%	2.51%
Glass/frit				
BOD	surface water	8.20e-06	< 0.01%	

Table 2-40. CRT manufacturing stage water outputs (wastewaters and pollutants)

Material	Disposition	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process group				
Chloride ions	surface water	1.01e+00	21.73%	
Chromium	surface water	8.20e-08	<0.01%	
COD	surface water	8.20e-06	<0.01%	
Dissolved solids	surface water	3.62e+00	77.84%	
Fluorides (F-)	surface water	2.93e-03	0.06%	
Iron	surface water	2.77e-03	0.06%	
Lead	surface water	4.34e-05	<0.01%	
Nickel	surface water	8.20e-08	<0.01%	
Nitrates/nitrites	surface water	3.95e-06	<0.01%	
Oil & grease	surface water	7.22e-03	0.16%	
Suspended solids	surface water	7.23e-03	0.16%	
Total		4.65e+00	100.00%	97.45%
РШВ				
Copper (+1 & +2)	treatment	9.71e-05	85.71%	
Lead cmpds	treatment	1.62e-05	14.29%	
Total		1.13e-04	100.00%	0.00%
Japanese grid				
Sulfate ion (-4)	surface water	8.72e-04	97.46%	
Suspended solids	surface water	2.27e-05	2.54%	
Total		8.94e-04	100.00%	0.02%
U.S. grid				
Sulfate ion (-4)	treatment	1.09e-03	97.46%	
Suspended solids	treatment	2.84e-05	2.54%	
Total		1.12e-03	100.00%	0.02%
Fuels				
Acids (H+)	surface water	2.50e-09	<0.01%	
Adsorbable organic halides	surface water	2.27e-15	<0.01%	
Aluminum (+3)	surface water	8.62e-10	<0.01%	
Ammonia ions	surface water	1.01e-07	0.10%	
Aromatic hydrocarbons	surface water	5.33e-13	<0.01%	
Barium cmpds	surface water	1.71e-12	<0.01%	
BOD	surface water	1.21e-06	1.22%	
Cadmium cmpds	surface water	1.78e-15	<0.01%	
Chloride ions	surface water	3.56e-05	36.07%	
Chromium (III)	surface water	3.31e-10	<0.01%	
Chromium (VI)	surface water	3.31e-10	<0.01%	
COD	surface water	9.80e-06	9.92%	
Copper (+1 & +2)	surface water	3.55e-14	<0.01%	
Cyanide (-1)	surface water	2.49e-15	<0.01%	
Dissolved organics	surface water	6.62e-09	<0.01%	
Dissolved solids	surface water	2.21e-07	0.22%	
Fluorides (F-)	surface water	1.80e-08	0.02%	

Table 2-40. CRT manufacturing stage water outputs (wastewaters and pollutants)

Material	Disposition	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process group		,		
Halogenated matter (organic)	surface water	7.11e-16	<0.01%	
Hydrocarbons, remaining unspeciated	surface water	2.61e-09	<0.01%	
Iron (+2 & +3)	surface water	3.79e-11	<0.01%	
Lead cmpds	surface water	7.11e-15	<0.01%	
Mercury compounds	surface water	8.17e-18	<0.01%	
Metals, remaining unspeciated	surface water	6.27e-08	0.06%	
Nickel cmpds	surface water	3.55e-15	<0.01%	
Nitrate	surface water	4.53e-09	<0.01%	
Other nitrogen	surface water	9.59e-14	<0.01%	
Phenol	surface water	2.21e-08	0.02%	
Phosphates	surface water	2.02e-11	<0.01%	
Polycyclic aromatic hydrocarbons	surface water	8.53e-15	<0.01%	
Salts (unspecified)	surface water	9.83e-09	<0.01%	
Sodium (+1)	surface water	4.59e-05	46.44%	
Sulfate ion (-4)	surface water	4.26e-09	<0.01%	
Sulfide	surface water	1.38e-09	<0.01%	
Suspended solids	surface water	5.19e-06	5.26%	
TOCs	surface water	5.33e-12	<0.01%	
Toluene	surface water	7.82e-14	<0.01%	
Waste oil	surface water	6.26e-07	0.63%	
Zinc (+2)	surface water	1.58e-10	<0.01%	
Tota	1	9.88e-05	100.00%	0.00%
Grand Tota	1	4.77e+00		100.00%

Table 2-40. CRT manufacturing stage water outputs (wastewaters and pollutants)

Process group		functional unit)	group total	total
Tube				
Frit	landfill	2.99e-03	15.58%	
Lead sulfate cake	landfill	2.67e-05	0.14%	
Silica coat waste	treatment	2.86e-04	1.49%	
Slag and ash	landfill	2.47e-03	12.90%	
Slurry scrap (chromium-based)	landfill	8.62e-04	4.50%	
Spent solvent, unspecified	treatment	2.75e-04	1.43%	
Unspecified sludge	landfill	5.22e-03	27.26%	
Unspecified sludge	recycling/reuse	5.56e-03	29.03%	
Waste oxygenated solvents	treatment	9.48e-05	0.49%	
Waste water treatment (WWT) filters	landfill	3.40e-04	1.78%	
Tota	1	1.81e-02	100.00%	1.61%
Glass/frit				
Barium debris (D008 waste)	landfill	2.14e-04	0.59%	
Broken CRT glass	landfill	1.88e-03	5.17%	
Chrome debris (D007 waste)	treatment	1.47e-04	0.41%	
Chrome liquid waste (D007 waste)	recycling/reuse	9.80e-03	26.95%	
cinders from CRT glass mfg (70% PbO)	landfill	8.26e-03	22.71%	
CRT glass faceplate EP dust (Pb) (D008 waste)	landfill	1.03e-03	2.83%	
CRT glass funnel EP dust (Pb) (D008 waste)	recycling/reuse	5.01e-03	13 78%	
Hazardous sludge (Pb) (D008)	landfill	1 52e-03	4 17%	
Hydrofluoric acid	landfill	1 78e-03	4 89%	
Lead contaminated grit (D008 waste)	landfill	3.46e-05	0.10%	
Lead debris (D008 waste)	landfill	2.14e-04	0.59%	
sludge from CRT glass mfg (1% PbO)	landfill	8.77e-04	2.41%	
Waste acid (mostly 3% HCl solution)	recycling/reuse	3.93e-03	10.81%	
Waste Batch (Ba, Pb) (D008 waste)	landfill	1.41e-03	3.89%	
Waste finishing sludge (Pb) (D008	landfill	2 56e-04	0.70%	
Tota		3.640-02	100.00%	3 73%
PWR		5.040-02	100.0070	5.2570
General Hazardous Waste	treatment	1 24e-01	27.26%	
PWB-Decontaminating debris	treatment	1.55e-02	3 41%	
PWB-L ead contaminated waste oil	treatment	1.55e 02	2 56%	
PWB-Route dust	recycling/reuse	1.10e 02	2.564%	
PWB-Solder dross	recycling/reuse	6 70e-02	14 72%	
PWB-Waste cupric etchant	recycling/reuse	2.25e-01	49 42%	
Tota		4.55e-01	100.00%	40.49%
Fuels	1		100.0070	
Hazardous waste	landfill	6.15e-01		54.67%

Table 2-41. CRT manufacturing stage hazardous waste outputs (kg/functional unit)

Material	Disposition	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process group				
Grand Tota	1	1.12e+00		100.00%

Table 2-41. CRT manufacturing stage hazardous waste outputs (kg/functional unit)

Material	Disposition	Quantity(kg/ functional unit)	% of process group total	% of grand total
Process group				
Monitor Assembly				
Broken CRT glass	recycling/reuse	3.82e-01	75.07%	
Cables/wires	recycling/reuse	8.86e-03	1.74%	
Printed wiring board (PWB)	recycling/reuse	3.70e-02	7.28%	
Waste plastics from CRT monitor	recycling/reuse	8.09e-02	15.91%	
Total		5.09e-01	100.00%	0.63%
Tube				
Broken CRT glass	recycling/reuse	6.94e-01	37.43%	
Ferric chloride	recycling/reuse	3.69e-01	19.93%	
Sludge (aquadag)	landfill	2.22e-03	0.12%	
Sludge (phosphor)	landfill	4.31e-03	0.23%	
Spent solvents				
(toluene,xylene,dimethyl				
formamide, isopropyl alcohol)	recycling/reuse	4.17e-02	2.25%	
Unspecified sludge	recycling/reuse	1.26e-01	6.78%	
Waste alkali (cleaning caustic and alkali soda effluent)	recycling/reuse	2.12e-02	1 15%	
Waste metals unspecified	recycling/reuse	8 79e-02	4 74%	
Waste oil	recycling/reuse	1 43e-03	0.08%	
Waste oil	treatment	2.55e-03	0.14%	
Waste Plastic (packing material)	treatment	3.01e-02	1.63%	
Waste Plastic (styrene foam)	recycling/reuse	3.77e-03	0.20%	
Waste water treatment (WWT) sludge	landfill	8.43e-02	4.55%	
Waste water treatment (WWT) sludge	recycling/reuse	3.72e-01	20.06%	
Wastepaper	recycling/reuse	8.34e-03	0.45%	
Wood, average	landfill	4.94e-03	0.27%	
Total		1.85e+00	100.00%	2.28%
Glass/frit				
abrasive sludge	recycling/reuse	4.21e-02	11.34%	
acid absorbent	landfill	8.13e-05	0.02%	
blasting media	landfill	3.66e-04	0.10%	
Cobalt nitrate	treatment	6.10e-05	0.02%	
CRT glass, faceplate	landfill	2.43e-02	6.54%	
Diesel fuel	treatment	4.07e-05	0.01%	
Dust	treatment	3.43e-03	0.92%	
Nickel nitrate	treatment	6.10e-05	0.02%	
Oily rags & filter media	landfill	3.25e-04	0.09%	
Oily rags & filter media	recycling/reuse	4.07e-05	0.01%	
parts cleaner solvent	recycling/reuse	8.13e-05	0.02%	
Plating process sludge	landfill	3.28e-04	0.09%	
Potassium Carbonate	landfill	3.30e-03	0.89%	
sludge (calcium fluoride, CaF2)	recycling/reuse	1.75e-02	4.72%	

Table 2-42. CRT manufacturing stage solid waste outputs (kg/functional unit)

Material	Disposition	Quantity(kg/ functional unit)	% of process group total	% of grand total
Process group				
Sodium Carbonate	landfill	3.29e-03	0.89%	
Unspecified sludge	landfill	7.69e-03	2.07%	
Waste alkali, unspecified	treatment	4.21e-05	0.01%	
Waste oil	treatment	6.54e-03	1.76%	
Waste refractory	landfill	2.44e-03	0.66%	
Waste water treatment (WWT) sludge	landfill	2.59e-01	69.68%	
PM	landfill	5.33e-04	0.14%	
Total		3.72e-01	100.00%	0.46%
РШВ				
PWB-Drill dust	landfill	1.49e-02	0.36%	
Unspecified solid waste	recycling/reuse	4.33e-01	10.53%	
Unspecified solid waste	treatment	3.66e+00	89.11%	
Total		4.11e+00	100.00%	5.06%
Japanese grid				
Coal waste	landfill	6.48e-01	61.12%	
Dust/sludge	landfill	2.50e-01	23.59%	
Fly/bottom ash	landfill	1.62e-01	15.28%	
Total		1.06e+00	100.00%	1.31%
U.S. grid				
Coal waste	landfill	8.12e-01	61.10%	
Dust/sludge	landfill	3.14e-01	23.63%	
Fly/bottom ash	landfill	2.03e-01	15.27%	
Total		1.33e+00	100.00%	1.64%
Fuels				
Aluminum scrap	recycling/reuse	1.82e-04	<0.01%	
Aluminum scrap, Wabash 319	recycling/reuse	5.08e-07	<0.01%	
Bauxite residues	landfill	1.21e-02	0.02%	
FGD sludge	landfill	2.14e-01	0.30%	
Mineral waste	landfill	2.61e-03	<0.01%	
Mixed industrial (waste)	landfill	1.00e+00	1.39%	
Non toxic chemical waste				
(unspecified)	landfill	6.11e-04	<0.01%	
Slag and ash	landfill	6.66e+01	92.62%	
Slag and ash	recycling/reuse	6.85e-01	0.95%	
Unspecified solid waste (incinerated)	treatment	1.33e-02	0.02%	
Unspecified waste	landfill	3.38e+00	4.70%	
Total		7.19e+01	100.04%	88.62%
Grand Total		8.12e+01		100.00%

Table 2-42. CRT manufacturing stage solid waste outputs (kg/functional unit)

Tuble 2 10. Civi manufacturing stage fauloactive waste outputs									
Material	Disposition	Quantity (kg/ functional unit)	% of process group total	% of grand total					
Process group									
Japanese grid									
Low-level radioactive waste	landfill	1.10e-04	76.93%						
Uranium, depleted	landfill	3.31e-05	23.07%						
Т	otal	1.43e-04	100.00%	79.72%					
U.S. grid									
Low-level radioactive waste	landfill	2.80e-05	76.93%						
Uranium, depleted	landfill	8.41e-06	23.07%						
T	otal	3.65e-05	100.00%	20.28%					
Grand To	otal	1.80e-04		100.00%					

Table 2-43. CRT manufacturing stage radioactive waste outputs

Material	Disposition	Quantity (Bq/ functional unit)	Quantity (Bq/ unctional unit)% of process group total	
Process group				
Japanese grid				
Antimony-124 (isotope)	treatment	4.95e-01	< 0.01%	
Antimony-125 (isotope)	treatment	1.97e+00	< 0.01%	
Argon-41 (isotope)	air	1.00e+03	0.03%	
Barium-140 (isotope)	treatment	3.67e-02	< 0.01%	
Bromine-89 (isotope)	air	1.16e-04	<0.01%	
Bromine-90 (isotope)	air	4.72e-05	<0.01%	
Cesium-134 (isotope)	air	3.18e-03	<0.01%	
Cesium-134 (isotope)	treatment	1.32e+00	<0.01%	
Cesium-137 (isotope)	air	2.40e-02	<0.01%	
Cesium-137 (isotope)	treatment	1.99e+00	<0.01%	
Chromium-51 (isotope)	air	6.29e-02	<0.01%	
Chromium-51 (isotope)	treatment	2.39e+00	<0.01%	
Cobalt-57 (isotope)	air	1.69e-04	<0.01%	
Cobalt-57 (isotope)	treatment	5.78e-02	<0.01%	
Cobalt-58 (isotope)	air	2.16e-03	<0.01%	
Cobalt-58 (isotope)	treatment	2.35e+01	<0.01%	
Cobalt-60 (isotope)	air	1.62e-02	< 0.01%	
Cobalt-80 (isotope)	treatment	6.17e+00	< 0.01%	
Iodine-131 (isotope)	air	7.58e-02	< 0.01%	
Iodine-131 (isotope)	treatment	1.10e+00	< 0.01%	
Iodine-132 (isotope)	air	1.54e-02	<0.01%	
Iodine-132 (isotope)	treatment	4.17e-01	<0.01%	
Iodine-133 (isotope)	air	7.03e+01	<0.01%	
Iodine-133 (isotope)	treatment	4.72e-01	<0.01%	
Iodine-134 (isotope)	air	7.98e-02	<0.01%	
Iodine-135 (isotope)	air	4.01e-03	< 0.01%	
Iodine-135 (isotope)	treatment	3.38e-01	< 0.01%	
Iron-55 (isotope)	treatment	5.62e+00	< 0.01%	
Iron-59 (isotope)	treatment	2.88e-01	< 0.01%	
Krypton-85 (isotope)	air	1.66e+03	0.06%	
Krypton-85M (isotope)	air	8.06e+01	<0.01%	
Krypton-85M (isotope)	treatment	1.49e+00	< 0.01%	
Krypton-87 (isotope)	air	3.00e+01	< 0.01%	
Krypton-88 (isotope)	air	1.41e+02	< 0.01%	
Lanthanum-140 (isotope)	treatment	3.93e-02	< 0.01%	
Manganese-54 (isotope)	air	8.92e-04	<0.01%	
Manganese-54 (isotope)	treatment	1.57e+00	<0.01%	
Molybdenum-99 (isotope)	treatment	2.97e+06	98.42%	
Niobium-95 (isotope)	air	3.54e-05	<0.01%	
Niobium-95 (isotope)	treatment	4.05e-01	<0.01%	
Rubidium-88 (isotope)	air	3.29e-01	<0.01%	

Table 2-44. CRT manufacturing stage radioactivity outputs

Material	Disposition	Quantity (Bq/ functional unit)	% of process group total	% of grand total
Process group		, , , , , , , , , , , , , , , , , , ,		
Ruthenium-103 (isotope)	treatment	4.95e-02	<0.01%	
Silver-110M (isotope)	air	1.06e-06	<0.01%	
Silver-110M (isotope)	treatment	5.78e-01	<0.01%	
Sodium-24 (isotope)	treatment	8.80e-02	<0.01%	
Strontium-89 (isotope)	treatment	9.51e-02	<0.01%	
Strontium-90 (isotope)	treatment	2.24e-02	<0.01%	
Strontium-95 (isotope)	treatment	2.46e-01	<0.01%	
Sulfur-136 (isotope)	treatment	5.30e-02	<0.01%	
Technetium-99M (isotope)	air	4.75e-06	< 0.01%	
Technetium-99M (isotope)	treatment	3.45e-02	<0.01%	
Tin-113 (isotope)	treatment	5.46e-02	<0.01%	
Tritium-3 (isotope)	air	2.35e+03	0.08%	
Tritium-3 (isotope)	treatment	1.76e+04	0.58%	
Xenon-131M (isotope)	air	1.36e+02	<0.01%	
Xenon-131M (isotope)	treatment	1.81e+01	< 0.01%	
Xenon-133 (isotope)	air	1.30e+03	0.04%	
Xenon-133 (isotope)	treatment	2.78e+03	0.09%	
Xenon-133M (isotope)	air	1.96e+04	0.65%	
Xenon-133M (isotope)	treatment	2.28e+01	< 0.01%	
Xenon-135 (isotope)	air	7.39e+02	0.02%	
Xenon-135 (isotope)	treatment	2.07e+01	< 0.01%	
Xenon-135M (isotope)	air	1.41e+01	<0.01%	
Xenon-138 (isotope)	air	4.68e+01	<0.01%	
Zinc-85 (isotope)	treatment	2.65e-02	<0.01%	
Zirconium-95 (isotope)	air	9.16e-05	<0.01%	
Tota	ıl	3.01e+06	100.00%	79.70%
U.S. grid				
Antimony-124 (isotope)	treatment	1.26e-01	<0.01%	
Antimony-125 (isotope)	treatment	5.02e-01	<0.01%	
Argon-41 (isotope)	air	2.55e+02	0.03%	
Barium-140 (isotope)	treatment	9.33e-03	<0.01%	
Bromine-89 (isotope)	air	2.95e-05	< 0.01%	
Bromine-90 (isotope)	air	1.20e-05	< 0.01%	
Cesium-134 (isotope)	air	8.09e-04	< 0.01%	
Cesium-134 (isotope)	treatment	3.37e-01	<0.01%	
Cesium-136 (isotope)	treatment	1.44e-02	< 0.01%	
Cesium-137 (isotope)	air	6.11e-03	<0.01%	
Cesium-137 (isotope)	treatment	5.06e-01	<0.01%	
Chromium-51 (isotope)	air	1.60e-02	<0.01%	
Chromium-51 (isotope)	treatment	6.07e-01	< 0.01%	
Cobalt-57 (isotope)	air	4.30e-05	<0.01%	
Cobalt-57 (isotope)	treatment	1.47e-02	<0.01%	

 Table 2-44. CRT manufacturing stage radioactivity outputs

Material	Disposition	Quantity (Bq/ % of process group functional unit) total		% of grand total
Process group				
Cobalt-58 (isotope)	air	5.49e+01	<0.01%	
Cobalt-58 (isotope)	treatment	5.98e+00	<0.01%	
Cobalt-60 (isotope)	air	4.13e-03	<0.01%	
Cobalt-80 (isotope)	treatment	1.57e+00	<0.01%	
Iodine-131 (isotope)	air	1.93e-02	<0.01%	
Iodine-131 (isotope)	treatment	2.80e-01	<0.01%	
Iodine-132 (isotope)	air	3.92e-03	<0.01%	
Iodine-132 (isotope)	treatment	1.06e-01	<0.01%	
Iodine-133 (isotope)	air	1.79e+01	<0.01%	
Iodine-133 (isotope)	treatment	1.20e-01	<0.01%	
Iodine-134 (isotope)	air	2.03e-02	<0.01%	
Iodine-135 (isotope)	air	1.02e-03	<0.01%	
Iodine-135 (isotope)	treatment	8.60e-02	<0.01%	
Iron-55 (isotope)	treatment	1.43e+00	<0.01%	
Iron-59 (isotope)	treatment	7.34e-02	<0.01%	
Krypton-85 (isotope)	air	4.23e+02	0.06%	
Krypton-85M (isotope)	air	2.05e+01	<0.01%	
Krypton-85M (isotope)	treatment	3.78e-01	<0.01%	
Krypton-87 (isotope)	air	7.62e+00	< 0.01%	
Krypton-88 (isotope)	air	3.58e+01	<0.01%	
Lanthanum-140 (isotope)	treatment	9.99e-03	<0.01%	
Manganese-54 (isotope)	air	2.27e-04	<0.01%	
Manganese-54 (isotope)	treatment	4.00e-01	<0.01%	
Molybdenum-99 (isotope)	treatment	7.55e+05	98.41%	
Niobium-95 (isotope)	air	9.01e-06	<0.01%	
Niobium-95 (isotope)	treatment	1.03e-01	<0.01%	
Rubidium-88 (isotope)	air	8.37e-02	<0.01%	
Ruthenium-103 (isotope)	treatment	1.26e-02	<0.01%	
Silver-110M (isotope)	air	2.69e-07	<0.01%	
Silver-110M (isotope)	treatment	1.47e-01	<0.01%	
Sodium-24 (isotope)	treatment	2.24e-02	<0.01%	
Strontium-89 (isotope)	treatment	2.42e-02	<0.01%	
Strontium-90 (isotope)	treatment	5.69e-03	<0.01%	
Strontium-95 (isotope)	treatment	6.27e-02	<0.01%	
Sulfur-136 (isotope)	treatment	1.35e-02	<0.01%	
Technetium-99M (isotope)	air	1.21e-06	<0.01%	
Technetium-99M (isotope)	treatment	8.77e-03	<0.01%	
Tin-113 (isotope)	treatment	1.39e-02	<0.01%	
Tritium-3 (isotope)	air	5.98e+02	0.08%	
Tritium-3 (isotope)	treatment	4.47e+03	0.58%	
Xenon-131M (isotope)	air	3.45e+01	<0.01%	
Xenon-131M (isotope)	treatment	4.60e+00	< 0.01%	

Table 2-44. CRT manufacturing stage radioactivity outputs

Material	Disposition	Quantity (Bq/ functional unit)	% of process group total	% of grand total
Process group				
Xenon-133 (isotope)	air	4.98e+03	0.65%	
Xenon-133 (isotope)	treatment	7.07e+02	0.09%	
Xenon-133M (isotope)	air	3.31e+02	0.04%	
Xenon-133M (isotope)	treatment	5.79e+00	<0.01%	
Xenon-135 (isotope)	treatment	5.27e+00	<0.01%	
Xenon-135M (isotope)	air	3.59e+00	<0.01%	
Xenon-138 (isotope)	air	1.19e+01	<0.01%	
Zinc-85 (isotope)	treatment	6.75e-03	<0.01%	
Zirconium-95 (isotope)	air	2.33e-05	<0.01%	
Tot	al	7.67e+05	100.00%	20.27%
Fuels				
Radioactive substance (unspecified)	air	9.19e+02	99.08%	
Radioactive substance (unspecified)	surface water	8.52e+00	0.92%	
Tot	al	9.27e+02	100.00%	0.02%
Grand Tot	al	3.78e+06		100.00%

 Table 2-44. CRT manufacturing stage radioactivity outputs























Among the utility inputs, both fuels and electricity were greatest in the glass/frit manufacturing processes (Figures 2-26 and 2-27). The fuels, especially, are dominated by the glass/frit process group, representing 83% of the mass of all the fuels in the manufacturing life-cycle stage. It is LPG in this inventory that clearly dominates the fuel inputs at about 351 kg/functional unit (99% of the glass/frit fuel inputs) (Table 2-38). Water inputs are greatest in the fuel production processes, contributing 1,050 kg (or liter)/functional unit (Figure 2-28). Total energy use from manufacturing is shown in Figure 2-29. The glass/frit manufacturing process group contributes the greatest to the total energy impacts in the manufacturing stage. A sensitivity analysis will be conducted on the glass data and more details are provided in Section 2.7.3.

For outputs from the manufacturing stage, the mass of air emissions are dominated by fuel production (Figure 2-30). Individual material (pollutant) contributions for each process group are presented in Table 2-39. Wastewater outputs (i.e., the volume or mass of wastewater released) are also greatest from the fuel production processes (Figure 2-31; Table 2-40);

however, the mass of chemical pollutants in the wastewater is greatest from the glass/frit manufacturing process group (4.65 kg/functional unit) out of the total manufacturing stage mass of pollutants, which is 4.77 kg/functional unit (Figure 2-32; Table 2-40).

Hazardous wastes from the CRT manufacturing stage were a small portion of the overall hazardous wastes generated by mass. Nonetheless, for purposes of future manufacturing stage improvement assessments, Table 2-41 presents the individual material contributions for each manufacturing process group; and Figure 2-33 shows that fuels production contributes the most (0.62 kg/functional unit) hazardous waste to the manufacturing stage. Manufacturing solid wastes are not as small a portion of the total mass of solid wastes throughout the CRT life-cycle (42%) as hazardous wastes are, as was depicted in Figure 2-21. Fuels production is the greatest contributor to manufacturing-generated solid wastes (71.9 kg/functional unit), followed by PWB manufacturing (4.1 kg/functional unit) (Figure 2-34 and Table 2-43).

Radioactive waste and radioactivity are directly related to the electricity generation process and therefore, only the Japanese and U.S. electric grid processes generate these outputs (some small radioactivity outputs are generated by fuels production processes) in the manufacturing stage. (These outputs also occur in upstream processes that have an electric grid included in the inventory.) Tables 2-44 and 2-45 show that more radioactive wastes and radioactivity are from the Japanese grid. This is a result of more manuf acturing processes as modeled in this project being in Japan.

2.7.1.2 LCD inventory results

The LCD inventory is presented similar to the CRT inventory above. The total LCD inventory presented in Table 2-24 and Figures 2-5, 2-6, and 2-7 shows the inventory from all life-cycle stages combined. The totals by life-cycle stage are presented in Table 2-45, Figure 2-35, Figure 2-36, and Figure 2-37.

Inventory type	Upstream	Mfg	Use	EOL	Total	Units*		
Inputs								
Primary materials	2.35e+02	4.92e+01	8.01e+01	-2.19e+00	3.62e+02	kg		
Ancillary materials	1.06e+00	2.04e+02	1.29e+00	2.11e+00	2.08e+02	kg		
Water	2.63e+02	2.15e+03	4.25e+02	-1.80e+01	2.82e+03	kg		
Fuels	??	??	??	??	0.00e+00	kg (or L)		
Electricity	3.46e+01	3.16e+02	8.53e+02	1.62e-01	1.20e+03	MJ		
Total energy	6.33e+02	1.44e+03	8.53e+02	-8.44e+01	2.84e+03	MJ		
Outputs				_		_		
Air pollutants	1.12e+02	6.48e+01	1.68e+02	1.30e+00	3.46e+02	kg		
Wastewater	8.57e+00	3.12e+03	0	-2.41e+00	3.13e+03	kg		
Water pollutants	4.60e-01	1.23e+00	2.62e-02	-4.09e-02	1.68e+00	kg (or L)		
Hazardous waste	6.72e-03	4.64e+00	0	1.64e+00	6.29e+00	kg		
Solid waste	1.31e+01	1.26e+01	3.11e+01	-4.42e+00	5.23e+01	kg		
Radioactive waste	2.21e+01	3.14e+03	3.11e+01	-5.23e+00	3.19e+03	kg		
Radioactivity	1.20e+07	1.02e+07	1.79e+07	3.40e+03	4.01e+07	Bq		

Table 2-45. LCD inventory by life-cycle stage

*Per functional unit (i.e., one LCD monitor over its effective life)







Considering inputs, Figure 2-35 shows that of the inputs measured in mass, the water inputs constitute the majority of the inputs by mass for the entire life cycle, and most of the water inputs are in the manufacturing life-cycle stage. Details on each inventory type are provided below. When considering which life-cycle stage contributes most to an inventory category, the manufacturing stage has the largest inventory by mass for ancillary materials, fuels, and water inputs. Primary material inputs are dominated by the upstream stages while electricity inputs are dominated by the use stage. The total energy is dominated by the manufacturing life-cycle stage (Figure 2-36). Note that LPG production from glass manufacturing does not dominate much of the LCD inventory as it did for the CRT because of the smaller amount of glass used in the LCD compared to the CRT.

Of the outputs measured in mass (air emissions, wastewater, water pollutants and hazardous, solid, and radioactive waste), wastewater constitutes the greatest output (Table 2-46); however, wastewater alone is not used to calculate impacts. Instead, water pollutants are used to calculate impacts and therefore listed separately in the inventory. Of the remaining outputs measured in mass (i.e., air emissions, water pollutants and hazardous, solid and radioactive waste), which are shown in Figure 2-37, air emissions are the greatest contributor to the outputs. Note again, as mentioned for the CRT, that radioactivity is measured in Bequerels (Bq) and cannot be compared on the same scale.

Considering each output type and their contributions by life-cycle stage, the mass of water pollutants is greatest in the manufacturing life-cycle stage, due to the fuel production processes that support fuel consumption in the manufacturing processes being included in the manufacturing life-cycle stage. Wastewater and hazardous waste outputs are greatest in the manufacturing stage; air emissions, solid waste, radioactive waste, and radioactivity have the greatest contribution from the use stage. As with the CRT, all the output totals represented in Table 2-45 include outputs to all dispositions.

The tables and figures discussed above show the total inventories for particular input or output types by life-cycle stage. Tables in Appendix J list each material that contributes to those totals. Figures 2-38 through 2-50 show the total contribution by life-cycle stage, based on the entire input/output type-specific tables in Appendix J. Summary tables for the LCD (Tables 2-46 through 2-54), developed from the Tables in Appendix J, show the top contributing inventory items to each input or output type. Note that Table 2-48 includes input/output types that are classified together as utilities: water, fuel, electricity and total energy.

LCD Primary Inputs

Figure 2-38 shows that most LCD primary materials by mass are from the upstream lifecycle stages. The top 99.9% of the materials contributing to the total LCD primary input inventory are shown in Table 2-46. The largest material contributors are natural gas, coal and petroleum (a combined 89% of all the primary LCD inputs), which are used to generate electricity consumed throughout the life-cycle of the monitor. Most of the electricity consumed in the LCD life-cycle is in the manufacturing and use stages, as was seen in Figure 2-36. However, most of the natural gas primary material reported in the materials processing stage (229 kg/functional unit) is not used to generate electricity, but is an ancillary material in the LCD monitor/module manufacturing process. More detail on the processes that contribute greatest within the manufacturing stage will be presented after brief discussions of the life-cycle stage breakdowns for each inventory type. For the complete list of primary materials in the LCD inventory, the total mass, and the mass contribution of each life-cycle stage, see Appendix J, Table J-10.



























			(Kg/Tunetonur unit)			
Material	Upstream	Mfg	Use	EOL	Total	% of total
Natural gas (in ground)	2.29E+02	5.16E+00	0	-1.08E+00	2.33E+02	64.25%
Coal, average (in ground)	1.72E+00	8.03E+00	6.69E+01	1.27E-02	7.67E+01	21.15%
Petroleum (in ground)	7.09E-01	2.23E+01	1.42E+00	-1.00E+00	2.34E+01	6.45%
Natural gas	0	4.22E+00	5.22E+00	-5.75E-02	9.39E+00	2.59%
Assembled LCD monitor	0	0	6.50E+00	0	6.50E+00	1.79%
Iron (Fe, ore)	3.26E+00	0	0	0	3.26E+00	0.90%
Steel	0	2.53E+00	0	0	2.53E+00	0.70%
Assembled 15" LCD backlight unit	0	1.48E+00	0	0	1.48E+00	0.41%
LCD module	0	1.18E+00	0	0	1.18E+00	0.33%
Polycarbonate resin	0	5.16E-01	0	0	5.16E-01	0.14%
Bauxite (Al2O3, ore)	0	5.09E-01	0	0	5.09E-01	0.14%
Iron scrap	4.63E-01	0	0	0	4.63E-01	0.13%
LCD glass	0	4.52E-01	0	0	4.52E-01	0.12%
Poly(methyl methacrylate)	0	3.83E-01	0	0	3.83E-01	0.11%
15" LCD light guide	0	3.74E-01	0	0	3.74E-01	0.10%
PWB-laminate	0	3.74E-01	0	0	3.74E-01	0.10%
Printed wiring board (PWB)	0	3.74E-01	0	0	3.74E-01	0.10%
Styrene-butadiene copolymers	0	3.62E-01	0	0	3.62E-01	0.10%
PPE	0	3.00E-01	0	0	3.00E-01	0.08%
Cables/wires	0	2.34E-01	0	0	2.34E-01	0.06%
LCD front glass (with color filters)	0	1.78E-01	0	0	1.78E-01	0.05%
Aluminum (elemental)	0	1.34E-01	0	0	1.34E-01	0.04%
Sand	0	1.11E-01	0	0	1.11E-01	0.03%
Recycled LCD glass	0	9.54E-02	0	0	9.54E-02	0.03%

Table 2-46. T	op 99.9% of LCD	primary	y material in	puts (kg/functional	unit)
---------------	-----------------	---------	---------------	--------	---------------	-------

LCD Ancillary Inputs

As presented in Figure 2-39, the greatest mass of ancillary LCD inputs is in the manufacturing life-cycle stage at approximately 204 kg/functional unit. Table 2-47 shows that liquified natural gas (LNG) contributes about 93% to this total. It is in the LCD module/monitor manufacturing process where this large amount of LNG was reported as an ancillary material (to be discussed below). Note that this is separate from LNG reported as a fuel, and LNG as an ancillary material is not used to calculate energy impacts in the LCIA. Following LNG is nitrogen at about 3% and clay at less than 1% of the total ancillary materials by mass. Excluding LNG from the inventory, nitrogen constitutes about 50% and clay 14% of the total ancillary materials in the LCD life-cycle. The contributions from the manufacturing stage will be discussed in further detail below. See Table J-11 in Appendix J for the complete list of ancillary materials in the LCD inventory.

1		•	1	(0		,
Material	Upstream	Mfg	Use	EOL	Total	% of total
LNG	0	1.94e+02	0	0	1.94e+02	93.19%
Nitrogen	0	6.02e+00	0	0	6.02e+00	2.89%
Clay (in ground)	1.30e-03	0	0	1.69e+00	1.69e+00	0.81%
Limestone	0	1.08e-01	8.99e-01	1.71e-04	1.01e+00	0.48%
Sand (in ground)	9.55e-03	1.32e-03	0	5.60e-01	5.71e-01	0.27%
Sodium hydroxide	0	4.45e-01	0	0	4.45e-01	0.21%
Hydrogen	0	4.44e-01	0	0	4.44e-01	0.21%
Lime	0	4.74e-02	3.95e-01	7.49e-05	4.42e-01	0.21%
Sodium chloride (NaCl, in ground or sea)	4.37e-01	6.08e-04	0	1.08e-05	4.38e-01	0.21%
Limestone (CaCO3, in ground)	5.07e-01	5.49e-02	0	-1.55e-01	4.06e-01	0.20%
Isopropyl alcohol	0	3.49e-01	0	0	3.49e-01	0.17%
Sulfuric acid	0	3.25e-01	0	0	3.25e-01	0.16%

 Table 2-47. Top 99% of LCD ancillary material inputs (kg/functional unit)

LCD Utility Inputs

Utility inputs in the LCD life-cycle are presented in Table 2-48 and include fuel (kg/functional unit), electricity (MJ/functional unit), water inputs (kg or L/functional unit), and total energy (MJ/functional unit; a combination of fuel and electricity inputs). Table 2-48 and Figure 2-40 show that most fuels (26 kg/functional unit) are used in the manufacturing stage. This represents 67% of the total fuels. LPG (16.8 kg/functional unit) dominates the total fuel inputs at 44% of all the fuels in the LCD life-cycle. More detail as to the breakdown by process within the manufacturing stage will be presented below after each input/output type is discussed.

Electricity inputs are dominated by the use stage (853 MJ/functional unit), followed by the manufacturing stage (316 MJ/functional unit) (see Figure 2-41). When fuel energy and electrical energy are combined into a total energy input value, the overall energy from manufacturing exceeds that from the use stage (1,440 MJ/functional unit versus 853 MJ/functional unit). This is also depicted in Figure 2-42.

The other utility considered in Table 2-48 is water (Figure 2-43). Approximately 76% (2,150 L/functional unit) of the water inputs in the LCD life-cycle are from the manufacturing processes. The life-cycle stage contributing the next most to water inputs is the use stage at 15% (425 L/functional unit). The upstream contributes about 9% (263 L/functional unit). Table J-12 in Appendix J provides the complete list of inventory items for the LCD.

Material	Upstream	Mfg	Use	EOL	Total	% of total
Fuels (kg/functional unit):						
LPG	0	1.68E+01	0	1.38E-03	1.68E+01	43.63%
Natural gas (in ground)	1.03E+01	2.41E+00	0	-1.38E-01	1.26E+01	32.66%
LNG	0	3.22E+00	0	0	3.22E+00	8.34%
Coal, average (in ground)	2.49E+00	6.86E-01	0	-7.66e-03	3.17E+00	8.21%
Petroleum (in ground)	1.52E+00	4.84E-01	0	-3.81e-02	1.96e+00	5.08%
Kerosene	0	4.65E-01	0	0	4.65e-01	1.21%
Coal, lignite (in ground)	4.10E-01	0	0	0	4.10e-01	1.06%
Natural gas	0	1.16E+00	0	-8.61E-01	3.01e-01	0.78%
Steam	0	1.45E-01	0	0	1.45e-01	0.37%
Fuel oil #6	0	1.25E-01	0	0	1.25e-01	0.33%
Fuel oil #2	0	5.42E-02	0	0	5.42e-02	0.14%
Uranium (U, ore)	7.86E-05	1.15E-05	0	0	9.01e-05	<0.01%
Fuel oil #4	0	2.11E-01	0	-9.09e-01	-6.99e-01	-1.81%
Total fuels	1.47E+01	2.58e+01	0	-1.95E+00	3.86e+01	100.00%
Electricity (MJ/functional unit):						
Electricity	3.46e+01	3.16e+02	8.53e+02	1.62e-01	1.20e+03	
Water (kg or L/functional unit):						
Water	2.63E+02	2.15E+03	4.25E+02	-1.80E+01	2.82e+03	
Total energy (fuels and electricity,	MJ/function	al unit):				
Energy	6.33E+02	1.44E+03	8.53E+02	-8.44E+01	2.84E+03	

Table 2-48. LCD utility inputs

LCD Air Outputs

Air emissions from the LCD life-cycle are greatest (by mass) in the use stage as seen in Figure 2-44. This indicates that most air emissions by mass are from the generation of electricity used by consumers of the monitors. Forty-nine percent of the total life-cycle air emissions by mass (or about 168 kg/functional unit) are from the use stage. Carbon dioxide (CO_2) emissions from the use stage alone constitute about 166 kg/functional unit, or almost 48% of all air emissions by mass in the life-cycle and nearly 99% of the use stage air emissions. The remaining air emissions that contribute to the top 99.99% of air emissions are presented in Table 2-49 and the complete list of air emissions are presented alphabetically in Table J-13 in Appendix J. The appendix also provides life-cycle stage subtotals. The next largest air emissions, by life-cycle air emissions. All the air emissions in the inventory except for ethylacetate and methyl ethyl ketone from the manufacturing stage (a combined 1.36 x 10^{-4} kg/functional unit) were reported as being emitted directly to the air (see Appendix J, Table J-13). Only those materials directly released to the air are used to calculate impacts. This will be discussed further in Chapter 3.

Table 2-49. Top 99.9976 of ECD an emissions (kg/functional unit)							
Material	Disposition	Upstream	Mfg	Use	EOL	Total	% of total
Carbon dioxide	Air	1.07e+02	6.22e+01	1.66e+02	1.39e+00	3.36e+02	97.18%
Methane	Air	3.54e+00	1.22e-01	2.41e-01	-2.82e-02	3.87e+00	1.12%
Nitrogen oxides	Air	6.56e-01	7.62e-01	4.39e-01	-1.36e-02	1.84e+00	0.53%
Sulfur dioxide	Air	4.63e-02	2.95e-01	9.30e-01	2.96e-04	1.27e+00	0.37%
Tetramethyl ammonium	Air	0	6.43e-01	0	0	6.43e-01	0.19%
Other organics	Air	4.45e-01	1.35e-02	0	-2.41e-03	4.56e-01	0.13%
Carbon monoxide	Air	3.74e-01	3.85e-02	3.02e-02	-3.45e-03	4.39e-01	0.13%
Nitrogen fluoride	Air	0	2.45e-01	0	0	2.45e-01	0.07%
Nonmethane hydrocarbons	Air	2.07e-01	8.87e-03	0	-1.26e-03	2.15e-01	0.06%
Hydrochloric acid	Air	1.50e-03	6.58e-02	4.02e-02	-6.88e-04	1.07e-01	0.03%
Benzene	Air	8.85e-02	2.70e-03	4.36e-05	-4.82e-04	9.07e-02	0.03%
PM	Air	9.16e-02	6.99e-03	0	-1.24e-02	8.62e-02	0.02%
Ammonia	Air	1.12e-02	6.26e-02	0	-6.95e-05	7.37e-02	0.02%
Phosphine	Air	0	6.26e-02	0	0	6.26e-02	0.02%
Hydrofluoric acid	Air	2.27e-04	5.27e-02	5.02e-03	-1.47e-04	5.78e-02	0.02%
Sulfur oxides	Air	2.57e-02	4.07e-02	0	-1.93e-02	4.71e-02	0.01%
Unspecified LCD process	Air	0	4.49e-02	0	0	4.49e-02	0.01%
Cr-etchant, unspecified	Air	0	4.12e-02	0	0	4.12e-02	0.01%
Nitrogen dioxide	Air	3.08e-02	0	0	4.46e-04	3.12e-02	< 0.01%
PM-10	Air	3.45e-07	6.85e-03	2.16e-02	3.43e-06	2.84e-02	< 0.01%
Isopropyl alcohol	Air	0	1.78e-02	0	0	1.78e-02	<0.01%
Hydrocarbons, remaining	Air	9.30e-03	7.75e-03	0	-6.51e-04	1.64e-02	< 0.01%
Al-etchant, unspecified	Air	0	1.37e-02	0	0	1.37e-02	< 0.01%

 Table 2-49. Top 99.99% of LCD air emissions (kg/functional unit)

LCD Water Outputs

The volume (or mass) of wastewater released throughout the LCD life-cycle is approximately 3,128 L (kg) per functional unit. Approximately 19% of that is sent to treatment as opposed to direct discharge to surface water (81%; Figure 2-45). The mass of chemical pollutants within the wastewater streams were calculated separately from the total wastewater volume. The total mass of water pollutants released, presented by life-cycle stage are shown in Figure 2-46. Of the small amount of water pollutants released, the manufacturing life-cycle stage contributes the greatest with approximately 1.23 kg per functional unit. This is about 73% of all the water pollutants for the entire life-cycle. The upstream stages have the second greatest mass of water pollutants at nearly 0.46 kg/functional unit (27%). The use and EOL stages are small contributors, with the EOL being negative due to recovery processes within the EOL stage. To see the top 99% contributors to the water pollutant quantities, Table 2-50 reveals that chloride and sodium ions contribute nearly 61% to all the water pollutants in the life-cycle, mostly from the manufacturing and upstream stages. For the complete inventory, listing water pollutants alphabetically and by life-cycle stage, see Appendix J, Table J-14. Further details on the manufacturing stage will be provided later.

						/	
Material	Disposition	Upstream	Mfg	Use	EOL	Total	% of total
Chloride ions	surface water	2.33e-01	3.12e-01	0	-1.58e-02	5.29e-01	31.53%
Sodium (+1)	surface water	1.73e-01	3.41e-01	0	-2.03e-02	4.94e-01	29.42%
Dissolved solids	surface water	3.21e-03	1.75e-01	0	-5.69e-05	1.78e-01	10.63%
COD	surface water	7.67e-03	8.20e-02	0	-2.69e-03	8.70e-02	5.18%
Nitrogen	surface water	1.44e-05	7.98e-02	0	0	7.98e-02	4.76%
Suspended solids	surface water	4.98e-03	5.80e-02	0	-1.44e-03	6.15e-02	3.66%
BOD	treatment	0	5.74e-02	0	0	5.74e-02	3.42%
COD	treatment	0	3.90e-02	0	0	3.90e-02	2.33%
BOD	surface water	7.72e-04	2.79e-02	0	-3.18e-04	2.83e-02	1.69%
Sulfate ion (-4)	surface water	2.40e-02	2.94e-03	0	-1.20e-06	2.69e-02	1.60%
Sulfate ion (-4)	treatment	0	1.32e-04	2.55e-02	4.84e-06	2.57e-02	1.53%
Fluorides (F-)	surface water	5.14e-05	1.29e-02	0	-5.01e-06	1.30e-02	0.77%
Nitrogen	treatment	0	1.26e-02	0	0	1.26e-02	0.75%
Phosphorus (yellow or white)	treatment	0	6.91e-03	0	0	6.91e-03	0.41%
Waste oil	surface water	1.75e-03	4.87e-03	0	-2.06e-04	6.41e-03	0.38%
Suspended solids	treatment	0	5.60e-03	6.65e-04	1.26e-07	6.26e-03	0.37%
Phosphorus (yellow or white)	surface water	1.92e-06	4.33e-03	0	0	4.33e-03	0.26%
Colon bacillus (bacteria in		0		0	0		
large intestine)	surface water		3.89e-03			3.89e-03	0.23%
Oil & grease	treatment	0	3.61e-03	0	0	3.61e-03	0.21%

 Table 2-50. Top 99% of LCD water pollutant outputs (kg/functional unit)

LCD Hazardous Waste Outputs

The total mass of hazardous waste generated throughout the life-cycle of the LCD is about 6.29 kg/functional unit. Figure 2-47 shows that this is mostly from the manufacturing stage, which contributes 4.64 kg/functional unit, or almost 74%. The EOL stage hazardous waste outputs equal 1.64 kg/functional unit, or 26%. The disposition of the waste will be used to determine how impacts are calculated in Chapter 3. Only hazardous wastes sent to landfills are directly calculated as impacts, which will be presented and discussed in Chapter 3. Figure 2-47 shows what portion of hazardous wastes are landfilled, recycled/reused, treated or otherwise land-applied. Nearly all of the hazardous waste in the manufacturing stage (~99%) is recycled/reused or treated. Less than 1% of the hazardous waste from the manufacturing stage is landfilled. Nearly all the hazardous waste from the EOL stage is landfilled. Table 2-51 shows the top contributors to the LCD life-cycle. Note that multiple entries of a material are due to different dispositions for that material. See Table J-15 in Appendix J for the complete alphabetical inventory of hazardous waste outputs. Additional detail on the manufacturing stage are presented below.

Material	Disposition	Upstream	Mfg	Use	EOL	Total	% of
							total
Isopropyl alcohol	treatment	0	1.91E+00	0	0	1.91E+00	30.41%
EOL LCD Monitor, landfilled	landfill	0	0	0	1.64E+00	1.64E+00	26.13%
Waste acid (mainly HF)	recycling/reuse	0	5.69E-01	0	0	5.69E-01	9.04%
Thinner, unspecified	treatment	0	5.40E-01	0	0	5.40E-01	8.57%
Remover, unspecified	treatment	0	3.03E-01	0	0	3.03E-01	4.81%
Sodium sulfate	recycling/reuse	0	2.44E-01	0	0	2.44E-01	3.89%
Isopropyl alcohol	recycling/reuse	0	1.69E-01	0	0	1.69E-01	2.69%
Tetramethyl ammonium	recycling/reuse	0	1.42E-01	0	0	1.42E-01	2.26%
hydroxide							
Waste acid (mainly HF)	treatment	0	1.36E-01	0	0	1.36E-01	2.15%
PWB-Waste cupric etchant	recycling/reuse	0	9.93E-02	0	0	9.93E-02	1.58%
Remover, unspecified	recycling/reuse	0	8.84E-02	0	0	8.84E-02	1.40%
Hazardous waste, unspecified	treatment	0	6.16E-02	0	0	6.16E-02	0.98%
Rinse, unspecified	recycling/reuse	0	4.67E-02	0	0	4.67E-02	0.74%
Spent solvent (non-halogenated)	treatment	0	4.66E-02	0	0	4.66E-02	0.74%
Hazardous waste, unspecified	landfill	6.72E-03	2.97E-02	0	-1.05E-03	3.54E-02	0.56%
Waste acids, unspecified	recycling/reuse	0	3.24E-02	0	0	3.24E-02	0.52%
Unspecified sludge	land (other than	0	3.09E-02	0	0	3.09E-02	0.49%
PWB-Solder dross	recycling/reuse	0	2 96E-02	0	0	2 96E-02	0.47%
Acetone	treatment	0	2.77E-02	0	0	2.77E-02	0.44%
Waste solvent (photoresist)	treatment	0	2.17E-02	0	0	2.17E-02	0.35%
Waste solvent (photoresist)	recycling/reuse	0	2.05E-02	0	0	2.05E-02	0.33%
Spent solvent (with halogenated materials)	treatment	0	1.55E-02	0	0	1.55E-02	0.25%
Phosphoric acid	landfill	0	1.44E-02	0	0	1.44E-02	0.23%

 Table 2-51. Top 99% of LCD hazardous waste outputs (kg/functional unit)

LCD Solid Waste Outputs

Figure 2-48 shows that the use stage contributes the most amount (31 kg/functional unit) of solid waste by mass to the LCD life-cycle, and 100% of that waste is landfilled. The manufacturing stage contributes 12.6 kg/functional unit. In terms of mass, the greatest material contributors to the solid waste outputs for the LCD life-cycle are coal waste (~41%), followed by dust/sludge (16%). Most of this is from the generation of electricity in the use stage (Table 2-52). Note also that the mass of an LCD monitor that is assumed to be landfilled (0.89 kg/functional unit) is 1.7% of the total mass of solid waste in the LCD life-cycle. See Appendix J, Table J-16 for the complete LCD solid waste inventory. The manufacturing stage breakdown will be discussed at the end of this section.
Material	Disposition	Upstream	Mfg	Use	EOL	Total	% of
	_	_	_				total
Coal waste	landfill	0	2.28E+00	1.90E+01	3.60E-03	2.13E+01	40.64%
Dust/sludge	landfill	0	8.80E-01	7.34E+00	1.39E-03	8.23E+00	15.72%
Fly/bottom ash	landfill	0	5.70E-01	4.75E+00	9.01E-04	5.32E+00	10.16%
Unspecified solid waste	landfill	2.40E+00	0	0	-5.10E-01	1.89E+00	3.62%
Slag and ash	recycle/reuse	8.02E+00	3.40E+00	0	-9.67E+00	1.75E+00	3.35%
Unspecified solid waste	recycle/reuse	1.50E+00	2.11E-01	0	0	1.71E+00	3.27%
Unspecified solid waste	treatment	0	1.63E+00	0	0	1.63E+00	3.11%
Iron scrap	recycle/reuse	1.67E-01	0	0	1.10E+00	1.27E+00	2.42%
EOL LCD Monitor, incinerated	treatment	0	0	0	9.75E-01	9.75E-01	1.86%
EOL LCD Monitor, recycled	recycle/reuse	0	0	0	9.75E-01	9.75E-01	1.86%
EOL LCD Monitor, remanufactured	recycle/reuse	0	0	0	9.75E-01	9.75E-01	1.86%
EOL LCD Monitor, landfilled	landfill	0	0	0	8.94E-01	8.94E-01	1.71%
Unspecified sludge	recycle/reuse	0	8.46E-01	0	0	8.46E-01	1.62%
Waste LCD glass	recycle/reuse	0	7.20E-01	0	0	7.20E-01	1.38%
Unspecified waste	recycle/reuse	4.05E-01	1.72E-01	0	-9.83E-03	5.67E-01	1.08%
CARBON STEEL SCRAP	recycle/reuse	0	0	0	4.58E-01	4.58E-01	0.88%
Waste plastic from LCD modules	treatment	0	4.03E-01	0	0	4.03E-01	0.77%
Polycarbonate	recycle/reuse	0	0	0	3.90E-01	3.90E-01	0.75%
Waste alkali, unspecified	recycle/reuse	0	3.23E-01	0	0	3.23E-01	0.62%
Waste acid (containing F and detergents)	landfill	0	2.70E-01	0	0	2.70E-01	0.52%
Waste LCD glass	landfill	0	2.63E-01	0	0	2.63E-01	0.50%
Mineral waste	landfill	2.20E-01	1.26E-04	0	-4.46E-06	2.21E-01	0.42%
Mining waste	landfill	1.41E-01	0	0	-1.23E-06	1.41E-01	0.27%
Slag and ash	landfill	8.19E-02	3.49E-02	0	-1.99E-03	1.15E-01	0.22%
Waste acids, unspecified	treatment	0	1.05E-01	0	0	1.05E-01	0.20%
Mixed industrial (waste)	landfill	4.34E-02	4.83E-02	0	-1.35E-03	9.04E-02	0.17%
Waste alkali (color filter developer, unspecified)	recycle/reuse	0	8.91E-02	0	0	8.91E-02	0.17%

 Table 2-52. Top 99% of LCD solid waste outputs (kg/functional unit)

LCD Radioactive Waste Outputs

Radioactive waste outputs in the LCD inventory are limited to the electricity generation and steel production processes, with steel production processes accounting for only about 9% of the total. Therefore, radioactive wastes will be found wherever electricity is used in a process in the LCD life-cycle. Only very small amounts (approximately 0.0015 kg/functional unit) of radioactive waste are generated over the entire life-cycle of the LCD (Figure 2-49 and Table 2-53). As expected, the majority of this is linked to the use stage, where most electricity is used in the LCD life-cycle, followed by the manufacturing stage. Low-level radioactive waste (78%) and depleted uranium (21%) are most of the waste, with negligible amounts of highly radioactive waste and unspecified radioactive waste. The inventory of radioactive waste outputs is small,

2.7 SUMMARY OF LIFE-CYCLE INVENTORY RESULTS

and therefore, Table 2-53 lists all material outputs associated with radioactive waste, in descending order of quantity. Table J-17 in Appendix J lists these in alphabetical order.

Material	Disposition	Upstream	Mfg	Use	EOL	Total	% of	
							total	
Low-level radioactive waste	landfill	1.28E-04	3.74E-04	6.56E-04	1.24E-07	1.16E-03	78.49%	
Uranium, depleted	landfill	0	1.12E-04	1.97E-04	3.73E-08	3.09E-04	20.93%	
Radioactive waste (unspecified)	landfill	5.77E-06	0	0	0	5.77E-06	0.39%	
Highly radioactive waste (Class C)	landfill	2.72E-06	0	0	0	2.72E-06	0.18%	
Total radioactive wastes		1.37E-04	4.87E-04	8.52E-04	1.62E-07	1.48E-03	100.00%	

Fable 2-53.	LCD radioactive	waste outputs	s (kg/functional	unit)
--------------------	-----------------	---------------	------------------	-------

LCD Radioactivity Outputs

Radioactivity is also inventoried in this project as isotopes that are released to the environment. Radioactivity is measured in Bequerels and may be released to air, water, or land, or may also be treated. The quantity of radioactivity for each life-cycle stage and different dispositions are presented in Figure 2-50. Table 2-54 shows the top contributors to the total radioactivity outputs. Radioactivity outputs are associated with the generation of electricity and, therefore, the greatest quantity of radioactivity is from the use stage. Appendix J, Table J-18 lists the complete inventory.

Material	Disposition	Upstream	Mfg.	Use	EOL	Total	% of total
Molybdenum-99 (isotope)	treatment	0	1.01e+07	1.76e+07	3.35e+03	2.77e+07	69.09%
Plutonium-241 (isotope)	landfill	1.18e+07	0	0	0	1.18e+07	29.33%
Tritium-3 (isotope)	treatment	0	5.96e+04	1.04e+05	1.98e+01	1.64e+05	0.41%
Xenon-133 (isotope)	air	7.63e+02	4.98e+03	1.16e+05	2.21e+01	1.22e+05	0.30%
Xenon-133M (isotope)	air	0	6.59e+04	7.73e+03	1.47e+00	7.36e+04	0.18%
Plutonium-240 (isotope)	landfill	5.08e+04	0	0	0	5.08e+04	0.13%
Cesium-135 (isotope)	landfill	4.59e+04	0	0	0	4.59e+04	0.11%
Radon-222 (isotope)	air	4.30e+04	0	0	0	4.30e+04	0.11%
Plutonium-239 (isotope)	landfill	3.57e+04	0	0	0	3.57e+04	0.09%
Xenon-133 (isotope)	treatment	0	9.43e+03	1.65e+04	3.13e+00	2.60e+04	0.06%
Tritium-3 (isotope)	air	1.09e+02	7.98e+03	1.40e+04	2.65e+00	2.21e+04	0.05%
Krypton-85 (isotope)	air	5.45e+01	5.64e+03	9.88e+03	1.88e+00	1.56e+04	0.04%

 Table 2-54. Top 99.9% of LCD radioactivity outputs (Bq/functional unit)

LCD Manufacturing Stage

The inventory tables that show the specific materials in each inventory (i.e., those in Appendix J and Tables 2-46 through 2-54) are the sums of the materials from one or more processes within a life-cycle stage. To burrow down deeper into the data, the manufacturing stage inventory data are broken down by process or group of processes. Similar to the CRT analysis, groups of processes were combined where fewer than three companies provided data for a process or where confidentiality agreements precluded presenting individual process data (Table 2-55). Burrowing further into the contributing processes or process groups is necessary for future manufacturing improvement assessments.

Process group	Process(es) included
Monitor/module	panel/module manufacturing, monitor assembly
Panel components	polarizer manufacturing, patterning color filters on glass, liquid crystal manufacturing
LCD glass	LCD glass manufacturing
Backlight	backlight unit assembly, backlight light guide, cold cathode fluorescent lamp manufacturing
PWB	PWB manufacturing
Japanese grid	electricity generation - Japanese electric grid
U.S. grid	electricity generation - U.S. electric grid
Fuels	production of fuel oils #2, #4 and #6, LPG, and natural gas

 Table 2-55.
 LCD process groups

Tables 2-56 through 2-64 list the specific inventories for each process group in the manufacturing stage for each input and output type. Figures 2-51 through 2-61 graph the total inventories for each process group for each input and output type. As revealed in Table 2-45, ancillary material inputs, fuels, water and total energy inputs all were greatest in the manufacturing stage. Similarly, wastewater, water pollutant and hazardous waste outputs were also greatest in the manufacturing stage. Similar to the discussion for CRTs, the manufacturing stage inventories by process group, for all input and output types, are presented here to reveal more specifics about the inventory and to allow manufacturers to conduct improvement assessments.

Material	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group			
Module/Monitor			
1,4-butanolide	4.06e-04	0.01%	
1-methyl-2-pyrrolidinone	4.06e-04	0.04%	
2-(2-butoxyethoxy)-ethanol acetate	8.08e-06	<0.01%	
AlNd	2.97e-05	<0.01%	
Aluminum (elemental)	1.01e-01	1.34%	
Assembled 15" LCD backlight unit	1.48e+00	19.67%	
Cables/wires	2.30e-01	3.07%	
Glycol ethers	4.06e-04	0.01%	
Indium tin oxide	5.26e-04	0.01%	
LCD front glass (with color filters)	1.78e-01	2.38%	
LCD glass	2.16e-01	2.88%	
LCD material (confidential)	3.11e-04	<0.01%	
LCD module	1.18e+00	15.78%	
LCD spacers, unspecified	1.69e-05	<0.01%	
Liquid crystals, for 15" LCD	1.24e-03	0.02%	
Mild fiber	7.34e-07	<0.01%	
Molybdenum	1.78e-04	<0.01%	
MoW	9.09e-04	0.01%	
Polarizer	4.07e-02	0.54%	
Polycarbonate resin	4.01e-01	5.35%	
Polyimide alignment layer, unspecified	4.86e-04	0.01%	
PPE	3.00e-01	4.00%	
Printed wiring board (PWB)	3.74e-01	4.98%	
Solder (60% tin, 40% lead)	3.81e-02	0.51%	
Steel	2.50e+00	33.38%	
Styrene-butadiene copolymers	3.62e-01	4.82%	
Titanium	1.33e-04	<0.01%	
Triallyl isocyanurate	1.54e-05	<0.01%	
Triphenyl phosphate	9.25e-02	1.23%	
Unspecified LCD material	1.19e-04	<0.01%	
Total	7.50e+00	100.00%	15.26%
Panel Components			
3,4,5-trifluorobromobenzene	2.64e-04	0.08%	
3,4-difluorobromobenzene	3.65e-04	0.04%	
4-4(-propylcyclohexyl)cyclohexanone	2.18e-04	0.07%	
4-bromophenol	3.27e-04	0.10%	
4-ethylphenol	7.00e-05	0.02%	
4-pentylphenol	3.42e-04	0.11%	
4-propionylphenol	1.94e-04	0.06%	
LCD glass	2.36e-01	74.75%	
Pigment color resist, unspecified	3.72e-02	11.80%	

 Table 2-56.
 LCD manufacturing stage primary material inputs

Material	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group			
Polyester adhesive	6.25e-04	0.20%	
Polyethylene terephthalate	3.14e-02	9.96%	
Polyvinyl alcohol	8.61e-03	2.73%	
Tota	3.16e-01	100.00%	0.64%
Glass mfg			
Barium Carbonate	1.37e-02	4.90%	
Glass, unspecified	2.28e-03	0.23%	
Potassium Carbonate	1.75e-02	6.24%	
Recycled LCD glass	9.54e-02	34.00%	
Sand	1.11e-01	39.64%	
Sodium Carbonate	2.26e-02	8.05%	
Strontium Carbonate	1.53e-02	5.47%	
Zircon Sand	2.51e-03	0.90%	
Tota	2.81e-01	99.42%	0.57%
Backlight			
15" LCD light guide	3.74e-01	37.18%	
Aluminum (elemental)	3.35e-02	3.35%	
Argon	3.53e-05	<0.01%	
Backlight lamp (CCFL)	1.94e-03	0.19%	
Cables/wires	3.43e-03	0.34%	
Glass, unspecified	4.14e-02	4.11%	
Mercury	3.99e-06	<0.01%	
Metals, remaining unspeciated	6.81e-04	0.07%	
Neon	6.31e-05	0.01%	
Poly(methyl methacrylate)	3.83e-01	38.12%	
Polycarbonate resin	1.14e-01	11.38%	
Polyethylene terephthalate	2.74e-02	2.72%	
Rubber, unspecified	6.01e-04	0.06%	
Steel	2.52e-02	2.50%	
Tota	1.01e+00	100.04%	2.04%
РШВ			
PWB-laminate	3.74e-01	94.35%	
Solder (63% tin; 37% lead)	2.24e-02	5.66%	
Tota	3.96e-01	100.00%	0.81%
Japanese Grid			
Coal, average (in ground)	7.69e+00	47.41%	
Natural gas	4.20e+00	25.89%	
Petroleum (in ground)	4.33e+00	26.69%	
Uranium, yellowcake	1.02e-03	0.01%	
Tota	1.62e+01	100.00%	32.96%
U.S. Grid			
Coal, average (in ground)	3.47e-01	90.97%	

Table 2-56. LCD manufacturing stage primary material inputs

Material	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group			
Natural gas	2.71e-02	7.10%	
Petroleum (in ground)	7.36e-03	1.93%	
Uranium, yellowcake	9.39e-06	<0.01%	
Total	3.82e-01	100.01%	0.78%
Fuel Production			
Natural gas (in ground)	5.16e+00	22.33%	
Petroleum (in ground)	1.79e+01	77.67%	
Total	2.31e+01	100.00%	46.94%
Grand Total	4.92e+01		100.00%

Table 2-56. LCD manufacturing stage primary material inputs

Material	Quantity(kg/ functional unit)	% of process group total	% of grand total
Process Group			
Module/Monitor			
1,4-butanolide	4.04e-05	6.91%	
1-Methoxy-2-propanol	1.10e-02	100.00%	
2-(2-butoxyethoxy)-ethanol	3.04e-02	0.02%	
2,2,4-trimethylpentane	1.52e-05	< 0.01%	
2-ethoxyl ethylacetate	1.78e-03	< 0.01%	
Acetic acid	6.40e-03	< 0.01%	
Acetone	1.03e-02	0.01%	
Al-etchant, unspecified	5.88e-03	<0.01%	
Aluminum sulfate	1.05e-01	0.05%	
Ammonia	1.55e-02	0.01%	
Ammonium bifluoride	2.36e-03	< 0.01%	
Ammonium fluoride	1.14e-02	0.01%	
Ammonium hydroxide	5.15e-06	< 0.01%	
Argon	7.87e-03	< 0.01%	
Calcium hydroxide	1.39e-01	0.07%	
Carbon dioxide	3.74e-05	<0.01%	
Chlorine	1.55e-02	0.01%	
Cleaner, unspecified	1.47e-04	<0.01%	
Cresol-formaldehyde resin	8.29e-04	<0.01%	
Cr-etchant, unspecified	1.77e-02	0.01%	
Cyclohexane	2.03e-05	<0.01%	
Dimethylsulfoxide	6.63e-02	0.03%	
Ethanol	1.35e-02	0.01%	
Ethanol amine	7.85e-02	0.04%	
Ferric chloride	8.92e-03	<0.01%	
Fluorocarbon resin	3.38e-06	<0.01%	
Flux, unspecified	7.35e-05	<0.01%	
Glycol ethers	2.12e-02	0.01%	
Helium	6.18e-04	<0.01%	
Hexamethyldisilizane	2.58e-04	<0.01%	
Hydrochloric acid	4.31e-02	0.02%	
Hydrofluoric acid	4.21e-02	0.02%	
Hydrogen	4.44e-01	0.22%	
Hydrogen peroxide	1.47e-04	<0.01%	
Isopropyl alcohol	3.49e-01	0.17%	
ITO etchant, unspecified	2.94e-03	<0.01%	
Krypton	2.58e-05	<0.01%	
LNG	1.94e+02	95.80%	
Methyl ethyl ketone	7.35e-06	<0.01%	
Monosilane	1.12e-03	<0.01%	
N-Butylacetate	3.83e-02	0.02%	

Table 2-57	. LCD	manufacturing	stage ancillar	y material inputs

Material	Quantity(kg/ functional unit)	% of process group total	% of grand total
Process Group		6 1	
Nitric acid	1.24e-02	0.01%	
Nitrogen	5.90e+00	2.91%	
Nitrogen fluoride	1.08e-01	0.05%	
Nitrous oxide	1.36e-03	< 0.01%	
Oxygen	7.75e-03	< 0.01%	
Perfluoromethane	1.29e-03	< 0.01%	
Phosphine	2.69e-02	0.01%	
Phosphoric acid	3.95e-02	0.02%	
Photoresist, unspecified	1.38e-02	0.01%	
Polyaluminum chloride	6.40e-03	< 0.01%	
Polyethylene mono(nonylphenyl) ether glycol	3.40e-04	< 0.01%	
Polyimide, unspecified	2.94e-05	< 0.01%	
Propylene glycol	4.46e-03	< 0.01%	
Propylene glycol monomethyl ether acetate	1.56e-02	0.01%	
Rinse, unspecified	5.27e-02	0.03%	
Sodium dihydrogen phosphate dihydrate	4.06e-06	< 0.01%	
Sodium hydroxide	3.59e-01	0.18%	
Solder, unspecified	7.35e-05	< 0.01%	
Sulfur hexafluoride	1.62e-02	0.01%	
Sulfuric acid	2.29e-01	0.11%	
Surfactant, unspecified	1.09e-04	< 0.01%	
Synthetic resin, unspecified	6.57e-04	< 0.01%	
Tetramethyl ammonium hydroxide	1.29e-01	0.06%	
Unspecified LCD process material	2.58e-02	0.01%	
Water	6.88e-02	0.03%	
Xylene (mixed isomers)	1.57e-03	< 0.01%	
Total	2.03e+02	206.91%	99.48%
Panel Components			
Acetone	1.03e-02	3.19%	
Borax	9.13e-05	0.01%	
Carbon dioxide	4.82e-03	1.50%	
Cyclohexane	3.89e-03	1.21%	
Developing solution, unspecified	4.00e-02	12.45%	
Diluent, unspecified	8.27e-03	2.57%	
Ethanol	1.17e-02	3.65%	
Ethylacetate	9.68e-04	0.30%	
Exfoliation liquid, unspecified	1.43e-02	4.44%	
HCFC-225ca	1.37e-04	0.04%	
HCFC-225cb	1.37e-04	0.04%	
Heptane	1.03e-02	3.19%	
Hydrochloric acid	1.74e-03	0.54%	
Hydrogen	3.14e-06	<0.01%	

 Table 2-57. LCD manufacturing stage ancillary material inputs

Material	Quantity(kg/ functional unit)	% of process group total	% of grand total
Process Group			
Methyl ethyl ketone	4.84e-04	0.15%	
Nitric acid second cerium ammonium	1.13e-02	3.51%	
Nitrogen	1.17e-01	36.27%	
Orthoboric acid	7.30e-04	0.23%	
Perchloric acid	3.82e-03	1.19%	
Photoresist, unspecified	2.94e-03	0.91%	
Polyethylene terephthalate	3.20e-02	9.93%	
Sulfuric acid	1.58e-02	4.91%	
Tetrahydrofuran	3.82e-03	1.19%	
Toluene	2.75e-02	8.56%	
Total	3.22e-01	99.99%	0.16%
LCD Glass			
Aluminum Oxide	1.56e-03	17.30%	
Cerium Oxide	1.52e-04	1.68%	
Chromium Oxide	2.60e-06	0.03%	
Hydrofluoric acid	3.66e-03	40.61%	
Pumice	3.64e-03	40.37%	
Total	9.02e-03	100.00%	0.00%
Backlight			
Diethyl ether	9.28e-05	14.78%	
Ethanol	4.63e-05	7.36%	
Process material for backlight assembly	7.03e-05	11.19%	
Unspecified ancillary material	4.19e-04	66.67%	
Total	6.28e-04	100.00%	0.00%
РШВ			
Ammonium chloride	3.42e-02	6.68%	
Ammonium hydroxide	3.42e-02	3.42%	
Formaldehyde	2.91e-03	0.57%	
Glycol ethers	1.04e-02	2.03%	
Hydrochloric acid	8.46e-02	16.51%	
Hydrogen peroxide	1.37e-02	2.67%	
Nitric acid	5.99e-02	11.70%	
Polyethylene glycol	2.23e-02	4.34%	
Potassium hydroxide	1.88e-02	3.68%	
Potassium permanganate	5.14e-04	0.10%	
Potassium peroxymonosulfate	3.12e-02	6.08%	
PWB-solder mask solids	1.93e-02	3.76%	
Sodium Carbonate	1.42e-02	2.77%	
Sodium hydroxide	8.56e-02	16.71%	
Sulfuric acid	8.05e-02	15.72%	
Total	5.12e-01	96.74%	0.25%
Japanese Grid			

Table 2-57.	LCD	manufacturing	g stage	ancillary	y material inpu	ıts

Material	Quantity(kg/ functional unit)	% of process group total	% of grand total
Process Group			
Lime	4.53e-02	30.57%	
Limestone	1.03e-01	69.43%	
Total	1.48e-01	100.00%	0.07%
U.S. Grid			
Lime	2.05e-03	30.52%	
Limestone	4.66e-03	69.48%	
Total	6.71e-03	100.00%	0.00%
Fuel Production			
Bauxite (Al2O3, ore)	2.16e-03	3.66%	
Limestone (CaCO3, in ground)	5.49e-02	93.07%	
Sand (in ground)	1.32e-03	2.24%	
Sodium chloride (NaCl, in ground or in sea)	6.08e-04	1.03%	
Total	5.90e-02	100.00%	0.03%
Grand Total	2.04e+02		100.00%

Table 2-57. LCD manufacturing stage ancillary material inputs

Material	Quantity	% of process group total	% of grand total
Process group			
Fuels (kg/functional unit):			
Monitor/Module			
Fuel oil #4	2.11e-01	4.09%	
Kerosene	2.98e-01	5.77%	
LNG	3.22e+00	62.40%	
Liquified petroleum gas (LPG)	5.83e-01	11.30%	
Natural gas	8.48e-01	16.44%	
Total	5.16e+00	95.91%	20.13%
Panel Components			
Kerosene	1.68e-01	31.97%	
Natural gas	1.18e-07	<0.01%	
Steam (100 psig)	1.45e-01	27.56%	
Fuel oil #2	4.07e-04	0.08%	
Fuel oil #6	1.25e-01	23.91%	
Natural gas	8.64e-02	16.47%	
Total	5.25e-01	68.03%	2.05%
LCD glass			
Fuel oil #2	5.38e-02	0.33%	
Liquified petroleum gas (LPG)	1.62e+01	99.33%	
Natural gas	5.63e-02	0.34%	
Total	1.64e+01	100.00%	63.87%
Backlight			
LNG	4.17e-06	50.00%	
Natural gas	4.17e-06	50.00%	
Total	8.33e-06	200.00%	0.00%
РШВ			
Natural gas	??		ERR
Fuels			
Coal, average (in ground)	6.86e-01	19.19%	
Natural gas (in ground)	2.41e+00	67.27%	
Petroleum (in ground)	4.84e-01	13.54%	
Uranium (U, ore)	1.15e-05	<0.01%	
Total	3.58e+00	100.00%	13.96%
Grand Total	2.56e+01		100.00%
Electricity (MJ/functional unit):			
Monitor/Module	2.59e+02		81.80%
Panel Components	4.64e+01		14.70%
LCD glass	2.20e+00		0.70%
Backlight	4.46e+00		1.41%
РШВ	4.43e+00		1.40%
Total	3.16e+02		100.00%

Table 2-58. LCD manufacturing stage utility inputs

Material	Quantity	% of process group total	% of grand total
Process group			
Water (kg or L/functional unit):			
Monitor/Module	1.08e+03		49.96%
Panel Components	2.08e+02		9.66%
LCD glass	1.62e+00		0.08%
Backlight	1.92e+02		8.91%
РШВ	1.86e+01		0.86%
Japanese electric grid	1.49e+02		6.91%
U.S. electric grid	2.20e+00		0.10%
Fuels	5.07e+02		23.53%
Total	2.15e+03		100.00%
Total energy (fuels and electricity, MJ/functional unit):			
Monitor/Module	5.08e+02		35.36%
Panel Components	6.29e+01		4.38%
LCD glass	7.05e+02		49.03%
Backlight	4.46e+00		0.31%
PWB	1.21e+01		0.84%
Fuels	1.45e+02		10.09%
Total	1.44e+03		100.00%

Table 2-58. LCD manufacturing stage utility inputs

Material	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group		8	
Module/Monitor			
Acetic acid	1.36e-03	0.07%	
Acetone	1.86e-04	0.02%	
Al-etchant, unspecified	1.37e-02	0.75%	
Ammonia	6.23e-02	3.41%	
Argon	5.80e-03	0.32%	
Carbon dioxide	2.16e-03	0.12%	
Cr-etchant, unspecified	4.12e-02	2.25%	
Cyclohexane	4.85e-05	<0.01%	
Diethylene glycol	9.69e-05	0.01%	
Hexamethyldisilizane	1.37e-06	<0.01%	
Hydrochloric acid	6.06e-02	3.31%	
Hydrofluoric acid	5.21e-02	2.85%	
Hydrogen	1.33e-04	0.01%	
Isopropyl alcohol	1.78e-02	0.97%	
ITO etchant, unspecified	6.86e-03	0.38%	
Monosilane	1.54e-03	0.08%	
N-bromoacetamide	9.18e-03	0.50%	
Nitric acid	2.69e-04	0.01%	
Nitrogen fluoride	2.45e-01	13.43%	
Nitrogen oxides	5.48e-01	30.00%	
Phosphine	6.26e-02	3.43%	
Phosphoric acid	4.85e-05	<0.01%	
PM	1.10e-05	<0.01%	
Polyimide, unspecified	1.40e-04	0.01%	
Sulfur hexafluoride	7.30e-03	0.40%	
Sulfur oxides	1.12e-03	0.06%	
Tetramethyl ammonium hydroxide	6.43e-01	35.16%	
Unspecified LCD process material	4.49e-02	2.46%	
Total	1.83e+00	100.00%	2.82%
Panel Components			
Carbon dioxide	4.82e-03	81.79%	
Ethylacetate	2.44e-06	<0.01%	
HCFC-225ca	1.40e-04	2.37%	
HCFC-225cb	1.40e-04	2.37%	
Heptane	7.77e-05	1.32%	
Hydrochloric acid	7.32e-06	0.12%	
Methyl ethyl ketone	1.35e-04	2.29%	
Nitrogen oxides	4.11e-04	6.98%	
Nonmethane hydrocarbons, remaining unspeciated	7.77e-05	1.32%	
PM	2.74e-05	0.47%	
Toluene	5.44e-05	0.92%	

 Table 2-59. LCD manufacturing stage air outputs

Material	Quantity (kg/ functional unit)	% of process	% of grand
Process Group	runctional unity	group total	totai
Total	5.89e-03	100.00%	0.01%
LCD Glass		100.0070	000170
Carbon dioxide	1 30e-01	98 45%	
Carbon monoxide	2 07e-07	<0.01%	
Chromium	6.40e-09	< 0.01%	
Nitrogen oxides	2.04e-03	1.55%	
PM	5.06e-06	<0.01%	
Sulfur oxides	2.36e-06	<0.01%	
Total	1.32e-01	100.03%	0.20%
Backlight			
Diethyl ether	9.26e-05	0.31%	
Ethanol	4.63e-05	0.16%	
Nitrogen oxides	2.95e-02	99.29%	
Process material for backlight assembly	7.03e-05	0.24%	
Total	2.97e-02	100.00%	0.05%
РШВ			
Formaldehyde	1.71e-05		0.00%
Japanese Grid			
1,1,1-Trichloroethane	2.17e-07	<0.01%	
1,2-Dichloroethane	1.54e-07	<0.01%	
2,3,7,8-TCDD	5.65e-14	<0.01%	
2,3,7,8-TCDF	1.96e-13	<0.01%	
2,4-Dinitrotoluene	1.07e-09	<0.01%	
2-Chloroacetophenone	2.69e-08	<0.01%	
2-Methylnaphthalene	8.00e-10	<0.01%	
5-Methyl chrysene	8.46e-11	<0.01%	
Acenaphthene	1.45e-08	<0.01%	
Acenaphthylene	1.11e-09	<0.01%	
Acetaldehyde	2.19e-06	<0.01%	
Acetophenone	5.77e-08	<0.01%	
Acrolein	1.11e-06	<0.01%	
Anthracene	1.53e-09	<0.01%	
Antimony	3.20e-06	<0.01%	
Arsenic	2.36e-06	<0.01%	
Barium	1.74e-06	<0.01%	
Benzene	5.13e-06	<0.01%	
Benzo[a]anthracene	2.69e-09	<0.01%	
Benzo[a]pyrene	1.46e-10	<0.01%	
Benzo[b,j,k]fluoranthene	1.31e-09	<0.01%	
Benzo[g,h,i]perylene	1.45e-09	<0.01%	
Benzyl chloride	2.69e-06	<0.01%	
Beryllium	1.09e-07	<0.01%	

Material	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group			
Biphenyl	6.53e-09	<0.01%	
Bromoform	1.50e-07	<0.01%	
Bromomethane	6.15e-07	<0.01%	
Cadmium	4.74e-07	<0.01%	
Carbon dioxide	5.18e+01	99.14%	
Carbon disulfide	5.00e-07	<0.01%	
Carbon monoxide	9.43e-03	0.02%	
Chloride ions	2.07e-04	<0.01%	
Chlorobenzene	8.46e-08	<0.01%	
Chloroform	2.27e-07	<0.01%	
Chromium (III)	1.86e-06	<0.01%	
Chromium (VI)	4.51e-07	<0.01%	
Chrysene	1.80e-09	<0.01%	
Cobalt	3.98e-06	<0.01%	
Copper	1.07e-06	<0.01%	
Cumene hydroperoxide	2.04e-08	<0.01%	
Cyanide (-1)	9.61e-06	<0.01%	
Di(2-ethylhexyl)phthalate	2.81e-07	<0.01%	
Dibenzo[a,h]anthracene	9.95e-10	<0.01%	
Dichloromethane	1.11e-06	<0.01%	
Dimethyl sulfate	1.85e-07	<0.01%	
Dioxins, remaining unspeciated	2.51e-12	<0.01%	
Ethyl Chloride	1.61e-07	<0.01%	
Ethylbenzene	3.99e-07	<0.01%	
Ethylene dibromide	4.61e-09	<0.01%	
Fluoranthene	5.88e-09	<0.01%	
Fluorene	6.15e-09	<0.01%	
Fluorides (F-)	2.22e-05	<0.01%	
Formaldehyde	3.43e-05	<0.01%	
Furans, remaining unspeciated	4.00e-12	<0.01%	
Hexane	2.58e-07	<0.01%	
Hydrochloric acid	4.61e-03	0.01%	
Hydrofluoric acid	5.77e-04	<0.01%	
Indeno(1,2,3-cd)pyrene	1.51e-09	<0.01%	
Isophorone	2.23e-06	< 0.01%	
Lead (Pb. ore)	1.48e-06	< 0.01%	
Magnesium	4.23e-05	< 0.01%	
Manganese (Mn. ore)	3.67e-06	< 0.01%	
Mercury	3.97e-07	<0.01%	
Methane	2.73e-04	<0.01%	
Methyl chloride	2.04e-06	<0.01%	

Material	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group	,	3 •	
Methyl ethyl ketone	1.50e-06	<0.01%	
Methyl hydrazine	6.54e-07	<0.01%	
Methyl methacrylate	7.69e-08	<0.01%	
Methyl tert-butyl ether	1.34e-07	<0.01%	
Molybdenum	5.20e-07	<0.01%	
Naphthalene	7.44e-07	<0.01%	
Nickel	4.95e-05	<0.01%	
Nitrogen oxides	1.37e-01	0.26%	
Nitrous oxide	3.76e-04	<0.01%	
o-xylene	6.49e-08	<0.01%	
Phenanthrene	1.75e-08	<0.01%	
Phenol	6.15e-08	<0.01%	
PM-10	6.72e-03	0.01%	
Propionaldehyde	1.46e-06	<0.01%	
Pyrene	4.24e-09	<0.01%	
Selenium	5.40e-06	<0.01%	
Styrene	9.61e-08	<0.01%	
Sulfur dioxide	2.90e-01	0.56%	
Tetrachloroethylene	1.65e-07	<0.01%	
TOCs, remaining unspeciated	6.63e-04	<0.01%	
Toluene	4.81e-06	<0.01%	
Vanadium	1.92e-05	<0.01%	
Vinyl acetate	2.92e-08	<0.01%	
Xylene (mixed isomers)	1.42e-07	<0.01%	
Zinc (elemental)	1.73e-05	<0.01%	
Total	5.22e+01	100.84%	80.58%
U.S. Grid			
1,1,1-Trichloroethane	3.71e-09	<0.01%	
1,2-Dichloroethane	6.95e-09	<0.01%	
2,3,7,8-TCDD	2.48e-15	<0.01%	
2,3,7,8-TCDF	8.86e-15	<0.01%	
2,4-Dinitrotoluene	4.86e-11	<0.01%	
2-Chloroacetophenone	1.22e-09	<0.01%	
2-Methylnaphthalene	5.16e-12	<0.01%	
5-Methyl chrysene	3.82e-12	<0.01%	
Acenaphthene	1.10e-10	<0.01%	
Acenaphthylene	4.37e-11	<0.01%	
Acetaldehyde	9.90e-08	<0.01%	
Acetophenone	2.60e-09	<0.01%	

Material	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group			
Acrolein	5.04e-08	<0.01%	
Anthracene	3.77e-11	<0.01%	
Antimony	8.45e-09	<0.01%	
Arsenic	7.26e-08	<0.01%	
Barium	3.98e-09	<0.01%	
Benzene	2.26e-07	<0.01%	
Benzo[a]anthracene	1.80e-11	<0.01%	
Benzo[a]pyrene	6.60e-12	<0.01%	
Benzo[b,j,k]fluoranthene	2.06e-11	<0.01%	
Benzo[g,h,i]perylene	6.98e-12	<0.01%	
Benzyl chloride	1.22e-07	<0.01%	
Beryllium	3.69e-09	<0.01%	
Biphenyl	2.95e-10	<0.01%	
Bromoform	6.77e-09	<0.01%	
Bromomethane	2.78e-08	<0.01%	
Cadmium	9.33e-09	<0.01%	
Carbon dioxide	8.61e-01	98.98%	
Carbon disulfide	2.26e-08	<0.01%	
Carbon monoxide	1.57e-04	0.02%	
Chloride ions	3.52e-07	<0.01%	
Chlorobenzene	3.82e-09	<0.01%	
Chloroform	1.02e-08	<0.01%	
Chromium (III)	4.71e-08	<0.01%	
Chromium (VI)	1.40e-08	<0.01%	
Chrysene	1.98e-11	<0.01%	
Cobalt	2.35e-08	<0.01%	
Copper	1.93e-09	<0.01%	
Cumene	9.20e-10	<0.01%	
Cyanide (-1)	4.34e-07	<0.01%	
Di(2-ethylhexyl)phthalate	1.27e-08	<0.01%	
Dibenzo[a,h]anthracene	1.69e-12	<0.01%	
Dichloromethane	5.04e-08	<0.01%	
Dimethyl sulfate	8.33e-09	<0.01%	
Dioxins, remaining unspeciated	1.13e-13	<0.01%	
Ethyl Chloride	7.29e-09	<0.01%	
Ethylbenzene	1.64e-08	<0.01%	
Ethylene dibromide	2.08e-10	< 0.01%	
Fluoranthene	1.30e-10	<0.01%	
Fluorene	1.63e-10	<0.01%	

 Table 2-59. LCD manufacturing stage air outputs

Material	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group		8	
Fluoride	3.78e-08	<0.01%	
Formaldehyde	1.64e-07	< 0.01%	
Furans, remaining unspeciated	1.80e-13	< 0.01%	
Hexane	1.16e-08	< 0.01%	
Hydrochloric acid	2.08e-04	0.02%	
Hydrofluoric acid	2.60e-05	<0.01%	
Indeno(1,2,3-cd)pyrene	1.28e-11	<0.01%	
Isophorone	1.01e-07	<0.01%	
Lead	2.47e-08	<0.01%	
Magnesium	1.91e-06	<0.01%	
Manganese	8.83e-08	<0.01%	
Mercury	1.45e-08	<0.01%	
Methane	1.25e-03	0.14%	
Methyl chloride	9.20e-08	<0.01%	
Methyl ethyl ketone	6.77e-08	<0.01%	
Methyl hydrazine	2.95e-08	<0.01%	
Methyl methacrylate	3.47e-09	<0.01%	
Methyl tert-butyl ether	6.08e-09	<0.01%	
Molybdenum	1.13e-09	<0.01%	
Naphthalene	3.54e-09	<0.01%	
Nickel	1.33e-07	<0.01%	
Nitrogen oxides	2.28e-03	0.26%	
Nitrous oxide	6.58e-06	<0.01%	
o-xylene	1.10e-10	<0.01%	
Phenanthrene	4.85e-10	<0.01%	
Phenol	2.78e-09	<0.01%	
Phosphorus (yellow or white)	9.59e-09	<0.01%	
PM-10	1.12e-04	0.01%	
Propionaldehyde	6.60e-08	<0.01%	
Pyrene	6.45e-11	<0.01%	
Selenium	2.26e-07	<0.01%	
Styrene	4.34e-09	<0.01%	
Sulfur dioxide	4.83e-03	0.56%	
Tetrachloroethylene	7.47e-09	<0.01%	
TOCs, remaining unspeciated	1.12e-05	<0.01%	
Toluene	4.92e-08	<0.01%	
Vanadium	3.41e-08	<0.01%	
Vinyl acetate	1.32e-09	<0.01%	
Xylene (mixed isomers)	6.42e-09	<0.01%	

 Table 2-59. LCD manufacturing stage air outputs

Material	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group		3	
Zinc (elemental)	2.95e-08	<0.01%	
Total	8.70e-01	100.83%	1.34%
Fuel Production			
1,1,1-Trichloroethane	6.91e-09	<0.01%	
1,2-Dichloroethane	1.38e-08	<0.01%	
1,4-Dichlorobenzene	1.89e-08	<0.01%	
2,4-Dinitrotoluene	9.68e-11	<0.01%	
2-Chloroacetophenone	2.42e-09	<0.01%	
3-Methylcholanthrene	2.84e-11	<0.01%	
5-Methyl chrysene	7.60e-12	<0.01%	
Acenaphthene	2.77e-10	<0.01%	
Acenaphthylene	1.16e-10	<0.01%	
Acetaldehyde	1.97e-07	<0.01%	
Acetophenone	5.18e-09	<0.01%	
Acrolein	1.00e-07	<0.01%	
Aldehydes	8.98e-05	<0.01%	
Aluminum (elemental)	9.59e-07	<0.01%	
Ammonia	3.53e-04	<0.01%	
Anthracene	1.15e-10	<0.01%	
Antimony	1.15e-10	<0.01%	
Aromatic hydrocarbons	2.55e-09	<0.01%	
Arsenic	7.13e-07	<0.01%	
Barium	1.77e-08	<0.01%	
Benzene	2.70e-03	0.03%	
Benzo[a]anthracene	6.97e-11	<0.01%	
Benzo[a]pyrene	3.84e-11	<0.01%	
Benzo[b,j,k]fluoranthene	3.80e-11	<0.01%	
Benzo[b]fluoranthene	3.09e-11	<0.01%	
Benzo[g,h,i]perylene	3.21e-11	<0.01%	
Benzo[k]fluoranthene	3.09e-11	<0.01%	
Benzyl chloride	2.42e-07	<0.01%	
Beryllium	7.31e-08	<0.01%	
Biphenyl	5.88e-10	<0.01%	
Bromoform	1.35e-08	<0.01%	
Bromomethane	5.53e-08	<0.01%	
Butane	3.31e-05	<0.01%	
Cadmium	4.18e-08	<0.01%	
Calcium	8.31e-07	<0.01%	
Carbon dioxide	9.44e+00	97.19%	

Table 2-59.	LCD	manufacturing s	tage air	outputs
	100	manaraccarings	mge mi	ouputs

Material	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group			
Carbon disulfide	4.49e-08	<0.01%	
Carbon monoxide	2.89e-02	0.30%	
Chloride ions	1.75e-06	<0.01%	
Chlorine	1.01e-09	<0.01%	
Chlorobenzene	7.60e-09	<0.01%	
Chloroform	2.04e-08	<0.01%	
Chromium (III)	1.05e-06	<0.01%	
Chromium (VI)	1.05e-06	<0.01%	
Chrysene	6.70e-11	<0.01%	
Cobalt	1.27e-07	<0.01%	
Copper	7.98e-08	<0.01%	
Cumene	1.83e-09	<0.01%	
Cyanide (-1)	8.64e-07	<0.01%	
Di(2-ethylhexyl)phthalate	2.52e-08	<0.01%	
Dibenzo[a,h]anthracene	2.18e-11	<0.01%	
Dichloromethane	1.00e-07	<0.01%	
Dimethyl sulfate	1.66e-08	<0.01%	
Dimethylbenzanthracene	2.36e-10	<0.01%	
Dioxins, remaining unspeciated	5.60e-12	<0.01%	
Ethane	4.88e-05	<0.01%	
Ethyl Chloride	1.45e-08	<0.01%	
Ethylbenzene	3.28e-08	<0.01%	
Ethylene dibromide	4.15e-10	<0.01%	
Fluoranthene	3.01e-10	<0.01%	
Fluorene	3.66e-10	<0.01%	
Fluorides (F-)	1.98e-07	<0.01%	
Formaldehyde	5.78e-05	<0.01%	
Furans, remaining unspeciated	2.60e-11	<0.01%	
Halogenated hydrocarbons (unspecified)	1.42e-14	<0.01%	
HALON-1301	2.47e-11	<0.01%	
Hexane	2.84e-05	<0.01%	
Hydrocarbons, remaining unspeciated	7.75e-03	0.08%	
Hydrochloric acid	4.15e-04	<0.01%	
Hydrofluoric acid	5.18e-05	<0.01%	
Hydrogen sulfide	1.51e-04	<0.01%	
Indeno(1,2,3-cd)pyrene	5.31e-11	<0.01%	
Iron	1.85e-06	<0.01%	
Isophorone	2.00e-07	< 0.01%	
Lead	6.39e-07	< 0.01%	

 Table 2-59. LCD manufacturing stage air outputs

Material	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group			
Magnesium	3.80e-06	<0.01%	
Manganese	1.15e-06	<0.01%	
Mercury	4.55e-08	<0.01%	
Metals, remaining unspeciated	1.60e-08	<0.01%	
Methane	1.20e-01	1.24%	
Methyl chloride	1.83e-07	<0.01%	
Methyl ethyl ketone	1.35e-07	<0.01%	
Methyl hydrazine	5.88e-08	<0.01%	
Methyl methacrylate	6.91e-09	<0.01%	
Methyl tert-butyl ether	1.21e-08	<0.01%	
Molybdenum	9.92e-08	<0.01%	
Naphthalene	1.98e-08	<0.01%	
Nickel	6.06e-06	<0.01%	
Nitrogen oxides	4.27e-02	0.44%	
Nitrous oxide	8.13e-04	0.01%	
Nonmethane hydrocarbons, remaining unspeciated	8.79e-03	0.09%	
n-Propane	8.67e-08	<0.01%	
Other organics	1.35e-02	0.14%	
o-xylene	1.59e-07	<0.01%	
Pentane	4.10e-05	<0.01%	
Phenanthrene	1.22e-09	<0.01%	
Phenol	5.53e-09	<0.01%	
Phosphorus (yellow or white)	6.10e-07	<0.01%	
РМ	6.95e-03	0.07%	
PM-10	1.11e-05	<0.01%	
Polycyclic aromatic hydrocarbons	2.84e-12	<0.01%	
Propionaldehyde	1.31e-07	<0.01%	
Pyrene	2.00e-10	<0.01%	
Selenium	4.81e-07	<0.01%	
Silicon	8.31e-07	<0.01%	
Sodium	4.92e-06	<0.01%	
Styrene	8.64e-09	<0.01%	
Sulfur oxides	3.96e-02	0.41%	
Tetrachloroethylene	1.49e-08	<0.01%	
Toluene	2.63e-07	<0.01%	
Vanadium	1.30e-05	< 0.01%	
Vinyl acetate	2.63e-09	< 0.01%	
Zinc (elemental)	6.04e-07	<0.01%	
Total	9.72e+00	101.01%	15.00%

 Table 2-59. LCD manufacturing stage air outputs

2.7 SUMMARY OF LIFE-CYCLE INVENTORY RESULTS

	8 8	1	
Material	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group			
Grand T	otal 6.48e+01		100.00%

Material	Disposition	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group			-8	
WASTEWATER STREAMS				
Module/monitor	both	2.87e+03		91.80%
Panel components	both	3.53e-01		0.01%
LCD glass	surface water	1.67e+00		0.05%
Backlight	both	1.92e+02		6.15%
PWB	treatment	1.86e+01		0.60%
Fuels	surface water	4 33e+01		1.39%
Tota		3.12e+03		100.00%
WATER POLLUTANTS				
Module/monitor				
1,1,1-Trichloroethane	surface water	2.29e-08	< 0.01%	
Antimony	surface water	1.14e-07	< 0.01%	
Arsenic	surface water	1.14e-07	< 0.01%	
BOD	surface water	1.74e-02	6.69%	
BOD	treatment	5.05e-02	19.41%	
Boron	surface water	4.58e-06	< 0.01%	
Cadmium	surface water	1.14e-07	<0.01%	
Chromium	surface water	8.84e-06	<0.01%	
Chromium (VI)	surface water	2.29e-07	<0.01%	
COD	surface water	2.68e-03	1.03%	
COD	treatment	3.90e-02	14.99%	
Colon bacillus (bacteria in large intestine)	surface water	3.89e-03	1.50%	
Copper	surface water	9.18e-07	<0.01%	
Cyanide (-1)	surface water	3.66e-06	<0.01%	
Cyanide (-1)	treatment	6.67e-07	<0.01%	
Dissolved solids	surface water	7.55e-03	2.90%	
Fluorides (F-)	surface water	1.28e-02	4.91%	
Fluorides (F-)	treatment	2.40e-04	0.09%	
Hexane	surface water	5.88e-04	0.23%	
Iron	surface water	2.63e-06	<0.01%	
Lead	surface water	6.17e-06	<0.01%	
Manganese	surface water	2.29e-07	<0.01%	
Mercury	surface water	9.69e-08	<0.01%	
Nickel	surface water	2.29e-07	<0.01%	
Nitrogen	surface water	7.93e-02	30.45%	
Nitrogen	treatment	1.16e-02	4.45%	
Oil & grease	surface water	2.02e-04	0.08%	
Oil & grease	treatment	3.53e-03	1.35%	
Organic phosphorus, unspecified	surface water	2.29e-07	<0.01%	
Phenol	surface water	2.29e-07	<0.01%	
Phosphorus (yellow or white)	surface water	4.31e-03	1.65%	
Phosphorus (yellow or white)	treatment	6.91e-03	2.65%	

Material	Disposition	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group				
Polychlorinated biphenyls	surface water	1.14e-08	< 0.01%	
Suspended solids	surface water	1.55e-02	5.96%	
Suspended solids	treatment	4.26e-03	1.64%	
Tetrachloroethylene	surface water	2.29e-08	< 0.01%	
Tin	surface water	4.58e-07	< 0.01%	
Trichloroethylene	surface water	2.29e-08	< 0.01%	
Zinc (elemental)	surface water	2.63e-06	< 0.01%	
Total		2.60e-01	100.21%	21.11%
Panel components				
BOD	surface water	1.34e-03	15.44%	
BOD	treatment	3.89e-03	44.73%	
Borax	treatment	1.31e-06	0.02%	
COD	surface water	2.21e-03	25.45%	
Hydrochloric acid	treatment	3.29e-06	0.04%	
Nitrogen	surface water	5.71e-04	6.58%	
Orthoboric acid	treatment	1.31e-06	0.02%	
Phosphorus (yellow or white)	surface water	2.48e-05	0.29%	
Suspended solids	surface water	6.46e-04	7.43%	
Total		8.69e-03	100.00%	0.70%
LCD glass				
BOD	surface water	3.80e-07	< 0.01%	
Chloride ions	surface water	4.68e-02	21.73%	
Chromium	surface water	3.80e-09	<0.01%	
COD	surface water	3.80e-07	<0.01%	
Dissolved solids	surface water	1.68e-01	77.84%	
Fluorides (F-)	surface water	1.36e-04	0.06%	
Iron	surface water	1.28e-04	0.06%	
Lead	surface water	2.01e-06	<0.01%	
Nickel	surface water	3.80e-09	<0.01%	
Nitrate	surface water	1.83e-07	<0.01%	
Oil & grease	surface water	3.34e-04	0.16%	
Suspended solids	surface water	3.35e-04	0.16%	
Total		2.15e-01	100.00%	17.46%
Backlight				
BOD	treatment	3.00e-03	54.50%	
Iron	treatment	8.33e-05	1.51%	
Lead	treatment	8.33e-07	0.02%	
Mercury	treatment	8.33e-08	< 0.01%	
Nickel	treatment	3.33e-06	0.06%	
Nitrogen	treatment	1.00e-03	18.17%	
Oil & grease	treatment	8.33e-05	1.51%	
Suspended solids	treatment	1.33e-03	24.22%	

Process Group Interference Box P total Total 5.50e-03 100.00% 0.45 PWB Interference Interference 0 Copper (+1 & +2) treatment 4.28e-05 85.71% Lead cmpds treatment 7.14e-06 14.29% Iteration of the surface water 5.00e-05 100.00% 0.00 Japanese grid Interference 100.00% 0.00 Sulfate ion (-4) surface water 2.93e-03 97.46%	Process Group To PWB r (+1 & +2) mpds To Japanese grid e ion (-4) nded solids To U.S. grid e ion (-4) nded solids
Total 5.50e-03 100.00% 0.45 PWB Copper (+1 & +2) treatment 4.28e-05 85.71% Lead cmpds treatment 7.14e-06 14.29% Total 5.00e-05 100.00% 0.00 Japanese grid Comparison 2.93e-03 97.46% Sunface water 2.93e-03 97.46% 2.54%	To PWB r (+1 & +2) mpds To Japanese grid 2 ion (-4) nded solids To U.S. grid 2 ion (-4) ided solids
PWB Iteatment 4.28e-05 85.71% Copper (+1 & +2) treatment 7.14e-06 14.29% Lead cmpds treatment 7.14e-06 14.29% Total 5.00e-05 100.00% 0.00 Japanese grid Iteration (-4) Surface water 2.93e-03 97.46% Surface under solide purface under solide 7.62e.05 2.54%	PWB r (+1 & +2) smpds To Japanese grid 2 ion (-4) uded solids To U.S. grid 2 ion (-4) uded solids
Imb treatment 4.28e-05 85.71% Copper (+1 & +2) treatment 7.14e-06 14.29% Lead cmpds treatment 7.14e-06 14.29% Total 5.00e-05 100.00% 0.00 Japanese grid Sulfate ion (-4) surface water 2.93e-03 97.46% Sunnended solids autocase 7.62e.05 2.54%	Twb r (+1 & +2) smpds To Japanese grid e ion (-4) nded solids To U.S. grid e ion (-4) nded solids
Copper (+1 & +2) areament 4.280-03 63.717/8 Lead cmpds treatment 7.14e-06 14.29% Total 5.00e-05 100.00% 0.00 Japanese grid	To Japanese grid ion (-4) ided solids U.S. grid ion (-4) ided solids
Total 5.00e-05 100.00% 0.00 Japanese grid 2.93e-03 97.46% 2.54%	To Japanese grid 2 ion (-4) uded solids To U.S. grid 2 ion (-4) uded solids
Japanese grid 3.00E-03 100.00% 0.00 Sulfate ion (-4) surface water 2.93e-03 97.46% Surported solids gurface water 7.62e-05 2.54%	Japanese grid e ion (-4) nded solids U.S. grid e ion (-4) nded solids
Sulfate ion (-4) surface water 2.93e-03 97.46% Surface water 7.62a, 05 2.54%	Japanese gria e ion (-4) ided solids To U.S. grid e ion (-4) ided solids
Surface water $2.950-05$ 97.4070	nded solids U.S. grid ion (-4) ided solids
INVSPENCED SOLIDS / NAME	To U.S. grid 2 ion (-4) ided solids
Suspended solids Surface water 7.050-05 2.0470 Total 3.01e-03 100.00% 0.24	U.S. grid
US grid	e ion (-4) nded solids
Sulfate ion (-4) treatment 1 32e-04 97 46%	nded solids
Suspended solids treatment 3.45e-06 2.54%	
Suspended solids Ireatment 5.450 00 2.5470 Total 1 36e.04 100 00% 0 01	To
	Fuels
Acids (H+) surface water $1.76e-0.8 < 0.01\%$	(H+)
Adsorbable organic balides Surface water $1.700-00$ $<0.01\%$	bable organic balides
Aluminum $(+3)$ surface water 6 88e-06 <0.01%	num (+3)
Ammonia jons Surface water 1 22a 02 0 1804	num (+3)
Arimonia ions Surface water 1.55t-05 0.1670	tie hydrocerbons
Parium ampda surface water 1.260.08 <0.01%	n ampda
DOD surface water 0.12a.02 1.220/	Tempas
$\begin{array}{c} BOD \\ Sufface water \\ Codmium camp dc \\ \hline \\ Codmium camp dc$	un onnda
Cadimum cmpds Surface water 1.42e-11 <0.01%	da iona
Chromium (III) surface water 2.05e-01 55.7776	ium (III)
Chromium (III) Surface water 3.49e-09 <0.01%	ium (III)
COD	
$\frac{\text{COD}}{\text{Sufface water}} = \frac{2.84a}{2.84a} \frac{10}{2.84a} = \frac{2.84a}{2.84a} = 2.$	r (+1 e-+2)
Copper $(+1 \& +2)$ Sufface water $2.84e-10$ $<0.01\%$ Conside (1) $= 5666666666666666666666666666666666666$	$\frac{1}{1} \left(\frac{+1}{4} \times \frac{+2}{4} \right)$
Cyanide (-1) Surface water 1.99e-11 <0.01%	
Dissolved organics Surface water 4.67e-08 <0.01%	ved organics
Elucrides (E) surface water 1760.06 <0.01%	
$\frac{1}{1000} = \frac{1}{1000} = 1$	Les (r-)
Hadogenated matter (organic) Surface water 5.07e-12 <0.01%	entropy remaining unspeciated
$\frac{1}{1000} = \frac{1}{1000} = 1$	-2 + +2
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	-2 & +3)
Marcurry compounds surface water 6.52 of 14 < 0.01%	mpus
Metels remaining unspeciated surface water 0.52e-14 <0.01%	ry compounds
Niekel empde	ample
Nitroto	cinpus
NurateSurface water $2.9/e-00$ $<0.01\%$ Other rites corr $= 7.6(-10)$ $= 0.01\%$;
Other introgen Surface water /.66e-10 <0.01% Dhanel surface water 1.75 ± 0.4 0.020/	nuogen
r nenor surface water 1./Se-04 0.02% Dheenhatee 2.922.09 <0.010/	

Table 2-60.	LCD ma	inufacturing	stage	water outpu	its

Material	Disposition	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group				
Polycyclic aromatic hydrocarbons	surface water	6.81e-11	< 0.01%	
Salts (unspecified)	surface water	7.84e-05	0.01%	
Sodium (+1)	surface water	3.41e-01	46.05%	
Sulfate ion (-4)	surface water	4.36e-06	<0.01%	
Sulfide	surface water	9.86e-09	<0.01%	
Suspended solids	surface water	4.14e-02	5.59%	
TOCs	surface water	4.25e-08	< 0.01%	
Toluene	surface water	6.24e-10	<0.01%	
Waste oil	surface water	4.87e-03	0.66%	
Zinc (+2)	surface water	1.40e-09	<0.01%	
Tota	al	7.40e-01	100.27%	60.02%
Grand Tot:	al	1.23e+00		100.00%

Table 2-60. LCD manufacturing stage water outputs

Material	Disposition	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group				
Monitor/module				
Mercury	recycling/reuse	2.00e-06	< 0.01%	
Waste metals, unspecified	recycling/reuse	1.17e-03	0.03%	
Waste acids, unspecified	recycling/reuse	3.24e-02	0.76%	
Waste acid (mainly HF)	recycling/reuse	5.69e-01	13.30%	
Waste acid (mainly HF)	treatment	1.36e-01	3.17%	
Unspecified sludge	land (other than landfill)	3.09e-02	0.72%	
Thinner, unspecified	treatment	5.40e-01	12.62%	
Tetramethyl ammonium hydroxide	recycling/reuse	1.42e-01	3.33%	
Sodium sulfate	recycling/reuse	2.44e-01	5.72%	
Rinse, unspecified	recycling/reuse	4.67e-02	1.09%	
Remover, unspecified	recycling/reuse	8.84e-02	2.07%	
Remover, unspecified	treatment	3.03e-01	7.08%	
Phosphoric acid	landfill	1.44e-02	0.34%	
Nitric acid	landfill	3.43e-04	0.01%	
Isopropyl alcohol	recvcling/reuse	1.69e-01	3.95%	
Isopropyl alcohol	treatment	1.91e+00	44.75%	
Ferric chloride	recycling/reuse	1.37e-02	0.32%	
Acetone	treatment	2.77e-02	0.65%	
Acetic acid	landfill	4.46e-03	0.10%	
Total		4.28e+00	100.00%	92.11%
Panel components				
Spent solvents (F003 waste)	treatment	2.74e-04	0.24%	
Flammable liquids (F003 waste)	treatment	9.13e-04	0.80%	
Acid waste (D002 waste)	treatment	1.19e-03	1.04%	
Spent solvent (with halogenated				
materials)	treatment	1.55e-02	13.63%	
Spent solvent (non-halogenated)	treatment	4.66e-02	40.89%	
HCFC-225cb	recycling/reuse	3.11e-05	0.03%	
HCFC-225ca	recycling/reuse	3.11e-05	0.03%	
Waste solvent (photoresist)	recycling/reuse	2.05e-02	18.00%	
Waste solvent (photoresist)	treatment	2.17e-02	19.05%	
Waste acid (chrome mixed acid)	recycling/reuse	7.18e-03	6.29%	
Total		1.14e-01	100.00%	2.46%
LCD glass				
Waste Batch (Ba, Pb) (D008 waste)	landfill	6.55e-05	8.18%	
Waste acid (mostly 3% HCl solution)	recycling/reuse	1.82e-04	22.74%	
Hydrofluoric acid	landfill	8.24e-05	10.29%	
Chrome liquid waste (D007 waste)	recycling/reuse	4.54e-04	56.70%	
Chrome debris (D007 waste)	treatment	6.83e-06	0.85%	
Barium debris (D008 waste)	landfill	9.91e-06	1.24%	
Tota		8.01e-04	100.00%	0.02%

Table 2-61. LCD manufacturing stage hazardous waste outputs

Material	Disposition	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group				
Backlight				
Hazardous waste, unspecified	recycling/reuse	1.42e-02	67.68%	
Hazardous waste, unspecified	treatment	6.80e-03	32.31%	
Waste glass, with mercury	landfill	1.05e-10	<0.01%	
Waste CCFL, with mercury	treatment	8.17e-10	<0.01%	
Waste CCFL, with lead	treatment	8.17e-08	<0.01%	
Silver	landfill	2.72e-09	<0.01%	
Chromium	landfill	1.52e-06	0.01%	
Tota	1	2.11e-02	100.00%	0.45%
PWB				
PWB-Waste cupric etchant	recycling/reuse	9.93e-02	49.42%	
PWB-Solder dross	recycling/reuse	2.96e-02	14.72%	
PWB-Route dust	recycling/reuse	5.31e-03	2.64%	
PWB-Lead contaminated waste oil	treatment	5.14e-03	2.56%	
PWB-Decontaminating debris	treatment	6.85e-03	3.41%	
Hazardous waste, unspecified	treatment	5.48e-02	27.26%	
Tota	1	2.01e-01	100.00%	4.33%
Fuels				
Hazardous waste, unspecified	landfill	2.97e-02		0.64%
Grand Tota	1	4.64e+00		100.00%

Table 2-61. LCD manufacturing stage hazardous waste outputs

	D:			
Material	Disposition	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group			group total	
Monitor/module				
Isopropyl alcohol	treatment	1.03e-02	0.33%	
LCD panel waste	landfill	2.43e-02	0.77%	
Printed wiring board (PWB)	landfill	7.50e-03	0.24%	
Remover, unspecified	treatment	3.09e-02	0.98%	
Unspecified sludge	recycling/reuse	8.46e-01	26.79%	
Unspecified sludge	treatment	5.73e-02	1.82%	
Unspecified solid waste	recvcling/reuse	2.02e-02	0.64%	
Waste acid (containing F and detergents)	landfill	2.70e-01	8.56%	
Waste acids, unspecified	treatment	1.05e-01	3.32%	
Waste alkali, unspecified	recvcling/reuse	3.23e-01	10.24%	
Waste LCD glass	landfill	2.06e-01	6.52%	
Waste LCD glass	recycling/reuse	7.20e-01	22.80%	
Waste metals, unspecified	recycling/reuse	2.93e-03	0.09%	
Waste oil	treatment	1.61e-02	0.51%	
Waste plastic from LCD modules	recycling/reuse	7.40e-02	2.35%	
Waste plastic from LCD modules	treatment	4.03e-01	12.77%	
Waste plastics from LCD monitor	landfill	4.05e-02	1.28%	
Tota	1	3.16e+00	100.00%	25.07%
Panel components				
Isopropyl alcohol	recycling/reuse	2.53e-02	11.75%	
Polyester resin	recycling/reuse	3.20e-02	14.84%	
Unspecified solid waste	treatment	1.10e-02	5.09%	
Used silica gel	landfill	6.22e-04	0.29%	
Waste alkali (color filter developer,				
unspecified)	recycling/reuse	8.91e-02	41.37%	
Waste LCD glass	landfill	5.74e-02	26.66%	
Tota	l	2.15e-01	100.00%	1.71%
LCD glass				
abrasive sludge	recycling/reuse	1.95e-03	32.61%	
acid absorbent	landfill	3.77e-06	0.06%	
blasting media	landfill	1.70e-05	0.28%	
Cinders from LCD glass mfg	landfill	3.83e-04	6.40%	
Cobalt nitrate	treatment	2.83e-06	0.05%	
Diesel fuel	treatment	1.88e-06	0.03%	
Dust	treatment	1.59e-04	2.65%	
LCD glass EP dust	landfill	4.77e-05	0.80%	
LCD glass EP dust	recycling/reuse	2.32e-04	3.88%	
LCD glass, unspecified	landfill	1.13e-03	18.83%	
Nickel nitrate	treatment	2.83e-06	0.05%	
Oily rags & filter media	landfill	1.51e-05	0.25%	
Oily rags & filter media	recycling/reuse	1.88e-06	0.03%	

Table 2-62. LCD manufacturing stage solid waste outputs

Material	Disposition	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group			<u>a</u> - • • · p • • • • • • •	
parts cleaner solvent	recycling/reuse	3.77e-06	0.06%	
Plating process sludge	landfill	1.52e-05	0.25%	
Potassium Carbonate	landfill	1.53e-04	2.55%	
sludge (calcium fluoride, CaF2)	recycling/reuse	8.13e-04	13.59%	
Sludge from LCD glass mfg	landfill	4.06e-05	0.68%	
Sodium Carbonate	landfill	1.53e-04	2.55%	
Unspecified sludge	landfill	3.56e-04	5.95%	
Waste alkali, unspecified	treatment	1.95e-06	0.03%	
Waste LCD glass	landfill	8.70e-05	1.45%	
Waste oil	treatment	3.03e-04	5.07%	
Waste refractory	landfill	1.13e-04	1.89%	
Total		5.98e-03	100.00%	0.05%
Backlight				
Broken CCFL	landfill	2.69e-07	<0.01%	
Cardboard	treatment	1.82e-05	0.34%	
Polyethylene, foamed	treatment	9.99e-04	18.50%	
Polyethylene/polypropylene waste	treatment	2.72e-03	50.45%	
Unspecified nonhazardous waste	recycling/reuse	1.26e-04	2.32%	
Waste backlight casing (PC)	landfill	1.46e-05	0.27%	
Waste backlight light guide (PMMA)	landfill	1.52e-03	28.12%	
Total		5.40e-03	100.00%	0.04%
РШВ				
PWB-Drill dust	landfill	6.59e-03	0.36%	
Unspecified solid waste	recycling/reuse	1.91e-01	10.53%	
Unspecified solid waste	treatment	1.62e+00	89.11%	
Total		1.81e+00	100.00%	14.40%
Japanese grid				
Coal waste	landfill	2.18e+00	61.12%	
Dust/sludge	landfill	8.42e-01	23.59%	
Fly/bottom ash	landfill	5.45e-01	15.28%	
Total		3.57e+00	100.00%	28.34%
U.S. grid				
Coal waste	landfill	9.85e-02	61.10%	
Dust/sludge	landfill	3.81e-02	23.63%	
Fly/bottom ash	landfill	2.46e-02	15.27%	
Total		1.61e-01	100.00%	1.28%
Fuels				
Aluminum scrap	recycling/reuse	8.77e-06	<0.01%	
Aluminum scrap, Wabash 319	recycling/reuse	2.37e-08	<0.01%	
Bauxite residues	landfill	5.87e-04	0.02%	
FGD sludge	landfill	1.09e-02	0.30%	
Mineral waste	landfill	1.26e-04	<0.01%	

Table 2-62 .	LCD manufacturing stage solid waste out	nuts
	ECD manufacturing stage sona waste out	Juis

Material	Disposition	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group				
Mixed industrial (waste)	landfill	4.83e-02	1.32%	
Non toxic chemical waste (unspecified)	landfill	2.95e-05	<0.01%	
Slag and ash	landfill	3.40e+00	92.69%	
Slag and ash	recycling/reuse	3.49e-02	0.95%	
Unspecified solid waste (incinerated)	treatment	6.44e-04	0.02%	
Unspecified waste	landfill	1.72e-01	4.70%	
Total		3.66e+00	100.00%	29.10%
Grand Total		1.26e+01		100.00%

Table 2-62. LCD manufacturing stage solid waste outputs

 Table 2-63. LCD manufacturing stage radioactive waste outputs

Material	Disposition	Quantity (kg/ functional unit)	% of process group total	% of grand total
Process Group				
Japanese grid				
Low-level radioactive waste	landfill	3.71e-04	76.93%	
Uranium, depleted	landfill	1.11e-04	23.07%	
Total		4.82e-04	100.00%	99.09%
U.S. grid				
Low-level radioactive waste	landfill	3.40e-06	76.93%	
Uranium, depleted	landfill	1.02e-06	23.07%	
Total		4.42e-06	100.00%	0.91%
Grand Total		4.87e-04		100.00%

Material	Disposition	Quantity (Bq/ functional unit)	% of process group total	% of grand total
Process Group				
Japanese grid				
Antimony-124 (isotope)	treatment	1.67e+00	<0.01%	
Antimony-125 (isotope)	treatment	6.63e+00	< 0.01%	
Argon-41 (isotope)	air	3.37e+03	0.03%	
Barium-140 (isotope)	treatment	1.23e-01	< 0.01%	
Bromine-89 (isotope)	air	3.90e-04	<0.01%	
Bromine-90 (isotope)	air	1.59e-04	<0.01%	
Cesium-134 (isotope)	air	1.07e-02	<0.01%	
Cesium-134 (isotope)	treatment	4.45e+00	<0.01%	
Cesium-137 (isotope)	air	8.08e-02	<0.01%	
Cesium-137 (isotope)	treatment	6.69e+00	<0.01%	
Chromium-51 (isotope)	air	2.11e-01	<0.01%	
Chromium-51 (isotope)	treatment	8.02e+00	<0.01%	
Cobalt-57 (isotope)	air	5.68e-04	<0.01%	
Cobalt-57 (isotope)	treatment	1.94e-01	<0.01%	
Cobalt-58 (isotope)	air	7.26e-03	<0.01%	
Cobalt-58 (isotope)	treatment	7.90e+01	<0.01%	
Cobalt-60 (isotope)	air	5.46e-02	<0.01%	
Cobalt-80 (isotope)	treatment	2.07e+01	<0.01%	
Iodine-131 (isotope)	air	2.55e-01	<0.01%	
Iodine-131 (isotope)	treatment	3.70e+00	<0.01%	
Iodine-132 (isotope)	air	5.18e-02	<0.01%	
Iodine-132 (isotope)	treatment	1.40e+00	<0.01%	
Iodine-133 (isotope)	air	2.37e+02	<0.01%	
Iodine-133 (isotope)	treatment	1.59e+00	<0.01%	
Iodine-134 (isotope)	air	2.68e-01	<0.01%	
Iodine-135 (isotope)	air	1.35e-02	<0.01%	
Iodine-135 (isotope)	treatment	1.14e+00	<0.01%	
Iron-55 (isotope)	treatment	1.89e+01	<0.01%	
Iron-59 (isotope)	treatment	9.70e-01	<0.01%	
Krypton-85 (isotope)	air	5.59e+03	0.06%	
Krypton-85M (isotope)	air	2.71e+02	<0.01%	
Krypton-85M (isotope)	treatment	5.00e+00	<0.01%	
Krypton-87 (isotope)	air	1.01e+02	<0.01%	
Krypton-88 (isotope)	air	4.73e+02	<0.01%	
Lanthanum-140 (isotope)	treatment	1.32e-01	<0.01%	
Manganese-54 (isotope)	air	3.00e-03	<0.01%	
Manganese-54 (isotope)	treatment	5.29e+00	<0.01%	
Molybdenum-99 (isotope)	treatment	9.98e+06	98.42%	
Niobium-95 (isotope)	air	1.19e-04	<0.01%	
Niobium-95 (isotope)	treatment	1.36e+00	<0.01%	
Rubidium-88 (isotope)	air	1.11e+00	<0.01%	

 Table 2-64. LCD manufacturing stage radioactivity outputs

Material	Disposition	Ouantity (Bg/	% of process	% of grand total
	P •••••••	functional unit)	group total	
Process Group				
Ruthenium-103 (isotope)	treatment	1.67e-01	<0.01%	
Silver-110M (isotope)	air	3.56e-06	< 0.01%	
Silver-110M (isotope)	treatment	1.94e+00	< 0.01%	
Sodium-24 (isotope)	treatment	2.96e-01	<0.01%	
Strontium-89 (isotope)	treatment	3.20e-01	< 0.01%	
Strontium-90 (isotope)	treatment	7.52e-02	< 0.01%	
Strontium-95 (isotope)	treatment	8.29e-01	< 0.01%	
Sulfur-136 (isotope)	treatment	1.78e-01	< 0.01%	
Technetium-99M (isotope)	air	1.60e-05	<0.01%	
Technetium-99M (isotope)	treatment	1.16e-01	< 0.01%	
Tin-113 (isotope)	treatment	1.84e-01	<0.01%	
Tritium-3 (isotope)	air	7.90e+03	0.08%	
Tritium-3 (isotope)	treatment	5.91e+04	0.58%	
Xenon-131M (isotope)	air	4.56e+02	<0.01%	
Xenon-131M (isotope)	treatment	6.08e+01	<0.01%	
Xenon-133 (isotope)	air	4.37e+03	0.04%	
Xenon-133 (isotope)	treatment	9.34e+03	0.09%	
Xenon-133M (isotope)	air	6.58e+04	0.65%	
Xenon-133M (isotope)	treatment	7.65e+01	<0.01%	
Xenon-135 (isotope)	air	2.48e+03	0.02%	
Xenon-135 (isotope)	treatment	6.97e+01	<0.01%	
Xenon-135M (isotope)	air	4.74e+01	<0.01%	
Xenon-138 (isotope)	air	1.57e+02	<0.01%	
Zinc-85 (isotope)	treatment	8.92e-02	<0.01%	
Zirconium-95 (isotope)	air	3.08e-04	<0.01%	
Total		1.01e+07	100.00%	99.09%
U.S. grid				
Antimony-124 (isotope)	treatment	1.53e-02	<0.01%	
Antimony-125 (isotope)	treatment	6.09e-02	<0.01%	
Argon-41 (isotope)	air	3.09e+01	0.03%	
Barium-140 (isotope)	treatment	1.13e-03	<0.01%	
Bromine-89 (isotope)	air	3.58e-06	<0.01%	
Bromine-90 (isotope)	air	1.45e-06	<0.01%	
Cesium-134 (isotope)	air	9.82e-05	<0.01%	
Cesium-134 (isotope)	treatment	4.09e-02	<0.01%	
Cesium-136 (isotope)	treatment	1.75e-03	<0.01%	
Cesium-137 (isotope)	air	7.41e-04	<0.01%	
Cesium-137 (isotope)	treatment	6.13e-02	<0.01%	
Chromium-51 (isotope)	air	1.94e-03	<0.01%	
Chromium-51 (isotope)	treatment	7.36e-02	<0.01%	
Cobalt-57 (isotope)	air	5.21e-06	<0.01%	
Cobalt-57 (isotope)	treatment	1.78e-03	<0.01%	

Table 2-64. LCD manufacturing stage radioactivity outputs

Material	Disposition	Quantity (Bq/ functional unit)	% of process group total	% of grand total
Process Group			a i	
Cobalt-58 (isotope)	air	6.65e+00	<0.01%	
Cobalt-58 (isotope)	treatment	7.25e-01	<0.01%	
Cobalt-60 (isotope)	air	5.01e-04	<0.01%	
Cobalt-80 (isotope)	treatment	1.90e-01	<0.01%	
Iodine-131 (isotope)	air	2.34e-03	<0.01%	
Iodine-131 (isotope)	treatment	3.39e-02	<0.01%	
Iodine-132 (isotope)	air	4.75e-04	<0.01%	
Iodine-132 (isotope)	treatment	1.28e-02	<0.01%	
Iodine-133 (isotope)	air	2.17e+00	<0.01%	
Iodine-133 (isotope)	treatment	1.45e-02	<0.01%	
Iodine-134 (isotope)	air	2.46e-03	<0.01%	
Iodine-135 (isotope)	air	1.24e-04	<0.01%	
Iodine-135 (isotope)	treatment	1.04e-02	<0.01%	
Iron-55 (isotope)	treatment	1.73e-01	<0.01%	
Iron-59 (isotope)	treatment	8.90e-03	<0.01%	
Krypton-85 (isotope)	air	5.13e+01	0.06%	
Krypton-85M (isotope)	air	2.48e+00	<0.01%	
Krypton-85M (isotope)	treatment	4.58e-02	<0.01%	
Krypton-87 (isotope)	air	9.25e-01	<0.01%	
Krypton-88 (isotope)	air	4.34e+00	<0.01%	
Lanthanum-140 (isotope)	treatment	1.21e-03	<0.01%	
Manganese-54 (isotope)	air	2.75e-05	<0.01%	
Manganese-54 (isotope)	treatment	4.85e-02	<0.01%	
Molybdenum-99 (isotope)	treatment	9.15e+04	98.41%	
Niobium-95 (isotope)	air	1.09e-06	<0.01%	
Niobium-95 (isotope)	treatment	1.25e-02	<0.01%	
Rubidium-88 (isotope)	air	1.02e-02	<0.01%	
Ruthenium-103 (isotope)	treatment	1.53e-03	<0.01%	
Silver-110M (isotope)	air	3.26e-08	<0.01%	
Silver-110M (isotope)	treatment	1.78e-02	<0.01%	
Sodium-24 (isotope)	treatment	2.72e-03	<0.01%	
Strontium-89 (isotope)	treatment	2.93e-03	<0.01%	
Strontium-90 (isotope)	treatment	6.90e-04	<0.01%	
Strontium-95 (isotope)	treatment	7.60e-03	<0.01%	
Sulfur-136 (isotope)	treatment	1.64e-03	<0.01%	
Technetium-99M (isotope)	air	1.47e-07	<0.01%	
Technetium-99M (isotope)	treatment	1.06e-03	<0.01%	
Tin-113 (isotope)	treatment	1.68e-03	<0.01%	
Tritium-3 (isotope)	air	7.25e+01	0.08%	
Tritium-3 (isotope)	treatment	5.42e+02	0.58%	
Xenon-131M (isotope)	air	4.18e+00	<0.01%	
Xenon-131M (isotope)	treatment	5.58e-01	<0.01%	

 Table 2-64. LCD manufacturing stage radioactivity outputs

Material	Disposition	Quantity (Bq/ functional unit)	% of process group total	% of grand total
Process Group				
Xenon-133 (isotope)	air	6.04e+02	0.65%	
Xenon-133 (isotope)	treatment	8.57e+01	0.09%	
Xenon-133M (isotope)	air	4.01e+01	0.04%	
Xenon-133M (isotope)	treatment	7.02e-01	<0.01%	
Xenon-135 (isotope)	air	2.28e+01	0.02%	
Xenon-135 (isotope)	treatment	6.39e-01	<0.01%	
Xenon-135M (isotope)	air	4.35e-01	<0.01%	
Xenon-138 (isotope)	air	1.44e+00	<0.01%	
Zinc-85 (isotope)	treatment	8.18e-04	< 0.01%	
Zirconium-95 (isotope)	air	2.82e-06	<0.01%	
Tot	al	9.30e+04	100.00%	0.91%
Fuels				
Radioactive substance (unspecified)	air	4.44e+01	99.08%	
Radioactive substance (unspecified)	surface water	4.11e-01	0.92%	
Tot	al	4.48e+01	100.00%	0.00%
Grand Tot	al	1.02e+07		100.00%

Table 2-64. LCD manufacturing stage radioactivity outputs






















Of the total 49 kg of primary materials per functional unit in the manufacturing stage, the fuels production contributes the greatest (23.1 kg/functional unit), followed by the Japanese electric grid (16.2 kg/functional unit) (Figure 2-51 and Table 2-56). Nearly all (203 out of 204 kg/functional unit) of the ancillary materials used during manufacturing are from the monitor/module process group (Figure 2-52 and Table 2-57). Liquified natural gas (LNG) constitutes about 96% (194 kg/functional unit) of the total ancillary materials in the monitor/module process group. As stated earlier, this is not used to calculate energy impacts.

Of the utility inputs, electricity and water inputs were greatest in the monitor/module manufacturing processes; fuels were greatest in the LCD glass manufacturing process group (Figures 2-53, 2-54, and 2-55). Sixty-three percent of the fuel inputs are from the LCD glass manufacturing process group. Within that group, the use of LPG clearly dominates at about 16.2 kg/functional unit (over 99% of the LCD glass fuel inputs) (Table 2-58).⁶ The monitor/module manufacturing fuel inputs are about 20% of the total manufacturing fuel inputs (5.2 kg/functional unit). About 259 MJ/functional unit of electricity in the LCD manufacturing stage are from the monitor/module processes⁷, or 82% of all manufacturing electricity (Figure 2-54). The total energy, which converts fuel mass into energy and adds that to the electrical energy, is greatest in the LCD glass manufacturing processes and contributes 705 MJ/functional unit to the 1,440 MJ/functional unit in the manufacturing process, contributing 1,080 kg (or liter)/functional unit (Figure 2-55), which is 50% of all the manufacturing water inputs. The fuels production and panel components process groups contribute about 24% and 10% to the water manufacturing inputs, respectively.

For outputs from the manufacturing stage, the mass of air emissions are dominated by the generation of electricity (Figure 2-57). Individual material (pollutant) contributions for each process group are presented in Table 2-59. Wastewater outputs (i.e., the volume or mass of wastewater released) are greatest for the monitor/module manufacturing processes (92%) (Figure 2-58), but only 21% of the chemical pollutants in the wastewater streams come from those processes (Figure 2-59). Table 2-60 shows the individual contributions from each material.

Hazardous wastes from the LCD manufacturing stage dominated over other life-cycle stages, and within the manufacturing stage, the monitor/module processes had the greatest hazardous waste outputs by mass (4.3 kg/functional unit) (Figure 2-60). The greatest contributors by mass are isopropyl alcohol (a total of 2.1 kg/functional unit, or 49% of all wastes from the monitor/module manufacturing processes) (Table 2-61). These wastes, however, are recycled and although they are a large portion of the inventory, they will not affect the impact assessment (to be presented in Chapter 3) as they are not directly released to the environment.

Solid wastes generated during the manufacturing stage were only about 21% of the overall solid wastes generated throughout the LCD life-cycle, as was shown earlier in

⁶ Note: An industry participant questioned the large fuel contribution reported here; however, further discussions with industry supported that no valid reason could justify removing these data. Glass energy inputs are evaluated in a sensitivity analysis (see Section 2.7.3 and 3.4).

⁷ This amount of electricity is consistent with the industry participant that expressed doubt in the large fuel energy contribution and subsequent overall energy use amount in module manufacturing.

Figure 2-48. Within the manufacturing stage, the fuels production, Japanese grid and monitor/module manufacturing process groups are all major contributors to the solid waste outputs (Figure 2-61). The individual material contributions are provided in Table 2-62.

Radioactive waste and radioactivity are directly related to the electricity generation process and therefore, only the Japanese and U.S. electric grid processes generate these outputs in the manufacturing stage. Tables 2-63 and 2-64 show that more radioactive wastes and radioactivity are from the Japanese electric grid. This is a result of more manufacturing processes being in Japan, as modeled in this project, as well as the greater fraction of nuclear power in the Japanese electric grid.

2.7.2 Relative Data Quality

Sections 2.2 through 2.6 (and associated appendices) discuss the data quality and data limitations for each life-cycle stage. Several factors contribute to the overall quality for an entire life-cycle stage. For example, the manufacturing stage includes several different processes that were collected from several different companies. The quality of one data set from one company may be very different from that of another company. Relative data quality estimates have been made for each life-cycle stage, including electricity generation, which is included in more than one life-cycle stage (Table 2-65). In addition, transportation data quality is listed in Table 2-65, although it has been excluded from the analysis due to the very low data quality.

Table 2-03. Relative data quality		
Life-cycle stage	Relative data quality	
Upstream	Moderate	
Manufacturing	Moderate to high	
Use	Moderate to high	
EOL	Low to moderate	
Electricity generation	High	
Transportation	Very low	

 Table 2-65. Relative data quality

2.7.3 Sensitivity Analyses

The inventory results presented above in Section 2.7.1 are the "baseline" results in this study. The baseline scenario includes the parameters/assumptions presented in the methodologies for the effective life scenario. However, due to assumptions and uncertainties in this LCA, as in any LCA, sensitivity analyses on the baseline results have been conducted. Four areas have been identified where sensitivity analyses were most warranted:

- use stage lifespan assumptions;
- glass manufacturing energy inputs;
- LCD monitor/module manufacturing energy inputs, and
- LCD EOL disposition assumptions.

Selected sensitivity analyses were chosen based on the data with either the greatest uncertainties or with a large uncertainty and a major contributor to the inventory results. The matrix in Table

2.7 SUMMARY OF LIFE-CYCLE INVENTORY RESULTS

2-66 shows the different sensitivity analyses or scenarios that are considered in the impact assessment results. Discussions of the sensitivity analyses for manufactured life (use stage), glass manufacturing energy inputs, LCD monitor manufacturing energy inputs, and LCD EOL inventories follow in this section. Complete inventories of each sensitivity analysis scenario are not presented; however, the effects determined in the LCIA results of the sensitivity analyses are shown in Chapter 3 (see Section 3.4).

Monitor type	Sensitivity analysis scenario
Baseline a	nalyses (for reference)
CRT	Effective life scenario with average glass energy inputs (all glass manufacturing energy data used)
LCD	<u>Effective life scenario</u> with average glass energy inputs (all glass manufacturing energy data used) and outliers in the LCD module manufacturing energy data removed
Sensitivity	analyses
CRT	<u>Manufactured life scenario</u> same as baseline except lifespan is based on manufactured life instead of effective life, which results in some revised functional equivalency calculations (see Section 2.7.3.1 below)
LCD	<u>Manufactured life scenario</u> same as baseline except lifespan is based on manufactured life, which results in some revised functional equivalency calculations (see Section 2.7.3.1 below)
CRT	Modified glass energy scenario same as baseline except comparatively high glass manufacturing energy inputs are removed
LCD	Modified glass energy scenario same as baseline except comparatively high glass manufacturing energy inputs are removed
LCD	Modified LCD module energy scenario same as baseline except LCD monitor/ module manufacturing energy outliers are included in the average
LCD	Modifed LCD EOL scenario same as baseline except LCD EOL dispositions are modified

Table 2-66	List of	sensitivity	analysi	scenarios
1 abic 2-00.	LIST OI	SCHSILIVILY	anary 513	o scenarios

2.7.3.1 Manufactured life scenario

To address uncertainties in the use stage lifespan assumptions, we applied the manufacturing life scenario to the CRT and LCD life-cycle profiles. (See Section 2.4 for a discussion of the product use stage and the differences in the "effective life" versus "manufactured life" life span assumptions.) Recall that the LCD manufactured life (45,000 hours) is 3.6 times greater than the CRT manufactured life (12,500 hours). In an LCA, comparisons are made based on functional equivalency. Therefore, if one monitor will operate for a longer period of time than another, impacts should be based on an equivalent use. Thus, based on equivalent use periods, under the manufactured life scenario 3.6 CRTs would need to be manufactured for every LCD. This was incorporated into the profile analysis for the comparative manufactured life LCA. Similarly, on average, 1.4 LCD backlights (which can be

cost-effectively replaced) will be needed during the manufactured lifetime of an LCD monitor. This was also incorporated into the profile. Thus, the following modifications were made:

- change the CRT electricity input in the use stage from 635 kWh (2,286 MJ) to 788 kWh (2,837 MJ); ⁸
- change the LCD electricity input in the use stage from 237 kWh (853 MJ) to 1,035 kWh (3,726 MJ);
- increase the manufacturing of CRTs by a factor of 3.6 to account for the functional equivalency of CRTs and LCDs. This was done by increasing the functional unit (22 kg CRT monitor) by a factor of 3.6, which equates to manufacturing 3.6 times more CRTs than in the baseline case; and
- increase the manufacturing of the LCD backlight lamp by a factor of 1.4 to account for the functional equivalency of LCDs and CRTs. This was done by increasing the backlight lamp mass (0.0023 kg), which is an input to the backlight unit assembly process, by a factor of 1.4.

Note that functional equivalency modification requires that the manufactured life scenario results be used only when comparing the CRT and LCD. These results cannot be accurately used to compare EL to ML for CRT or LCD. LCIA results of the sensitivity analysis are presented in Chapter 3.

2.7.3.2 Modified glass energy scenario

In the second case, the energy input values for CRT glass manufacturing (and consequently LCD glass manufacturing) were considered uncertain due to the large discrepancy in fuel and electricity values among the individual data sets. The baseline case uses the average of the data supplied and confirmed by the companies who supplied the data. However, because one set of data was significantly higher, that one set of data was removed from the profile for the sensitivity analysis. (A statistical evaluation of the glass manufacturing data for outliers could not be conducted because there were not enough data sets.)

In the baseline scenario, the averaged primary data from manufacturers of total energy to produce a kilogram of CRT or LCD glass was 1,560 MJ (433 kWh) of energy, with only 0.3% of that as electrical energy. The sensitivity analysis scenario assumes 16.3 MJ (4.5 kWh) per kilogram of glass produced, with approximately 30% as electrical energy. The majority of the fuel energy in the baseline scenario was from LPG. The energy consumption values can be compared to estimates for the entire glass industry, which includes the more prevalent general flat glass, as well as speciality glasses, such as CRT and LCD glass. In a report of the Glass Technology Roadmap Workshop (Energetics Inc. 1997), it was estimated that in practice, about 1.1 MJ of energy are required to melt a kilogram of glass; and electrical energy contributes approximately 13% of the total process energy requirements for CRT or LCD glass, it suggests the baseline data collected for this analysis may be inflated. Therefore, the sensitivity

⁸ This represents the electricity use for a 12,500 hour life span. This figure is then multiplied by a factor of 3.6 in the functional equivalency calculations (see third bullet, below).

analysis uses revised energy input values for glass production, but it is also not known whether these values represent the true energy requirements for CRT and LCD glass production. The sensitivity analysis is considered a lower bound of energy requirements for monitor glass production.

2.7.3.3 Modified LCD module energy scenario

LCD monitor/module manufacturing energy was another area of relatively large uncertainty and variability in the inventory data. The CDP received seven sets of LCD monitor/module manufacturing data from five companies in Japan and two in Korea. Of these, the manufacturing energy data from one company in Korea was incomplete and could not be used. For the remaining six data sets, total energy inputs ranged from 330 MJ to 7,310 MJ, with a mean and standard deviation of 2,269 MJ and 2,906 MJ, respectively. Given the wide variability in the data and large standard deviation, CDP researchers evaluated the data for outliers by breaking the total energy data points into quartile ranges. Minor outliers are then those within a certain range of multipliers beyond the middle 50 percent of the distribution. That is, the interquartile range (IQR) (i.e., the range of values representing the middle 50 percent) multiplied by 1.5 is the lower bound of the minor outlier and the IQR times three is the upper bound of the minor outlier. Anything beyond IQR times three is a major outlier. Using this approach, one data set was found to be a minor outlier and another was found to be a major outlier. These outliers were excluded from the averages used in the baseline analysis, but included in the averages used in the LCD monitor/module manufacturing energy sensitivity analysis.

Table 2-67 summarizes the energy inputs for the LCD monitor/module manufacturing process group under the baseline and modified LCD module energy scenarios. Note that total energy inputs are approximately 4.5 times lower under the baseline scenario. However, because of the different types of energy (fuel and electricity) employed by different manufacturers, the mean electric energy is higher for the baseline than the modified energy scenario.

2.7.3.4 LCD End-of-life dispositions

Finally, because very few desktop LCDs have reached their end of life, and usually only if they have been damaged in some way, very little is known about the percentage of LCDs that are remanufactured, recycled, landfilled or incinerated. In the baseline scenario, it was assumed that a certain proportion of monitors go to each EOL disposition. As the functional unit in this study is one monitor, we used those proportions to represent the probability that one monitor would go to the respective disposition. To address uncertainties in the allocation of disposition percentages, a sensitivity analysis was conducted with a different set of final disposition numbers. Details and assumptions for the sensitivity analysis are provided in Appendix I. Table 2-68 presents the distribution of LCD EOL dispositions assumed under the baseline and modifed EOL dispositions scenarios. LCIA results for the sensitivity analysis are presented in Section 3.4.

		Daschine and in	ounieu energy	scenarios	
	Total energy (MJ per monitor)	Electric energy (MJ per monitor)	Fuel energy (MJ per monitor)	% electric energy	% fuel energy
Baseline (exclude	es two outlier data	sets from the mear	is used in the inver	ntory)	
Range	333 to 934	199 to 359	48 to 695	25 to 88	12 to 75
Mean	508	259	249	60	40
Standard deviation	284	68	300	27	27
Modified Energy (includes two outlier data sets in the means used in the inventory)					
Range	333 to 7,317	125 to 359	48 to 7,146	2 to 88	12 to 98
Mean	2,274	222	2052	10	90
Standard deviation	2,906	79	2,956	36	36

Table 2-67. Energy inputs to the LCD monitor/module manufacturing process group under the baseline and modified energy scenarios

Table 2-68. Distribution of LCD EOL dispositions in the baseline and
modified EOL scenarios

Disposition	Baseline	Modified
Incineration	15%	15%
Recycling	15%	0%
Remanufacturing	15%	40%
Hazardous waste landfill	5%	5%
Solid waste landfill	50%	40%

REFERENCES

- Bocko, P. 2000. Personal communication with P. Bocko, Corning Incorporated and M.L. Socolof, University of Tennessee Center for Clean Products and Clean Technologies. October 5.
- CIA (Computer Industry Almanac, Inc.). 1997. Computer Industry Almanac. Arlington Heights, IL.
- Douglas, D. 1999. Personal communication with D. Douglas and E. Korman, Dell, and J.G. Overly, University of Tennessee Center for Clean Products and Clean Technologies. March.
- *Ecobilan*. 1999. DEAM, life-cycle inventory database developed by the *Ecobilan* Group, copyright 1999.
- EIA (Energy Information Administration). 1997. *Country Energy Data Report for 1997*. Web site available at http://www.eia.doe.gov.
- EIA. 1999a. Electric Power Monthly, February 1999 (With Data for November 1998). DOE/EIA-0226(99/02).
- EIA. 1999b. "1997 Residential Energy Consumption Survey," Housing Characteristics report. Web site available at http://www.eia.doe.gov/emeu/consumption/reports.html (under 'Residential Households').
- Energetics, Incorporated. 1997. Report of the Glass Technology Roadmap Workshop, April 24-25, 1997, Alexandria, Virginia. Sponsored by the U.S. Department of Energy, Corning, Incorporated, Fort Motor Company, Pilkington Libbey-Owens-Ford, Praxair, Inc., and Techneglas, Incorporated. Columbia, Maryland.
- EPA (Environmental Protection Agency). 1996. Compilation of Air Pollutant Emission Factors (AP-42), Fifth Edition, Volume I. AP-42. January.
- EPA. 1998. "Characterization of Municipal Solid Waste in the United States: 1997 Update." Report No. EPA530-R-98-007.
- Fanara, A. 1999. Personal communication with A. Fanara, U.S. Environmental Protection Agency (EPA) Energy Star Program, and J.G. Overly, University of Tennessee Center for Clean Products and Clean Technologies. July.
- FEPC (Federation of Electric Power Companies). 1996. "Electricity Review Japan: The FEPC." Web site available at http://www.fepc or jp/erj/data 03.html (1996 data).

- Goldwasser, S.M. 1999. "Notes on the Troubleshooting and Repair of Computer and Video Monitors." Web site available at http://plop.phys.cwru.edu/repairfaq/REPAIR/F monfaq.html.
- ISO (International Standards Organization). 1996. ISO 14040, Environmental Management -Life-cycle Assessment Principles and Framework. TC 2071 SC 5N 77. International Standards Organization, Paris.
- Lapp, J.C. 2000. Personal communication with J.C. Lapp, Corning Incorporated, and M.L. Socolof, University of Tennessee, Center for Clean Products and Clean Technologies. October 11 and October 23.
- Matthews, H. S., *et al.* 1997. "Disposition and End-of-Life Options for Personal Computers." Green Design Initiative Technical Report #97-10. Carnegie Mellon University, Pittsburgh, PA. July.
- McConnaughey, M. 1999. Personal communication with M. McConnaughey, ViewSonic, and J.G. Overly, University of Tennessee, Center for Clean Products and Clean Technologies. March.
- MCC (Microelectronics and Computer Technology Corporation). 1998. Computer Display Industry and Technology Profile. EPA 744-R-98-005. December.
- Nordman, B., M. A. Piette and K. Kinney. 1996. "Measured Energy Savings and Performance of Power-Managed Personal Computers and Monitors." Presented at the American Council for an Energy Efficient Economy Summer Study, Pacific Grove, CA. LBL-38057, UC-1600. August.
- NSC (National Safety Council). 1999. Electronic Product Recovery and Recycling Baseline Report: Recycling of Selected Electronic Products in the United States. Washington, DC. May.
- ORNL (Oak Ridge National Laboratory). 1994. Report 3: Estimating Externalities of Coal Fuel Cycles. Oak Ridge, TN. September.
- ORNL. 1995. Report 8: Estimating Externalities of Nuclear Fuel Cycles. Oak Ridge, TN. April.
- SETAC (Society of Environmental Toxicology and Chemistry). 1994. *Life-Cycle Assessment Data Quality: A Conceptual Framework*. SETAC and SETAC Foundation for Environmental Education, Inc., Washington, DC.
- Sharp, J. 2000. Personal communication with J. Sharp, Teredyne Corp., J.G. Overly, University of Tennessee, Center for Clean Products and Clean Technologies, and M.L. Socolof, University of Tennessee, Center for Clean Products and Clean Technologies. August

- Socolof, M.L, *et al.* 2000. "Preliminary Life-cycle Assessment Results for the Design for the Environment Computer Display Project." International Symposium on Electronics and the Environment, The Institute of Electrical and Electronics Engineers, Inc. (IEEE) Computer Society Conference Record, May 8-10, San Francisco, CA.
- TORNRC (The Oak Ridge National Recycle Center). 2001. Personal communication with D. McFarland and A. Bradley, TORNRC, and R. Dhingra, University of Tennessee, Center for Clean Products and Clean Technologies. April.
- Tsuda, D. 1999. Personal communication with D. Tsuda, Environmental Technologies and Strategies Department, Apple Computer, Inc., and M.L. Socolof, University of Tennessee, Center for Clean Products and Clean Technologies. July.
- Vorhees, G. 2000. Personal communication with G. Vorhees, Envirocycle, and M.L. Socolof, University of Tennessee, Center for Clean Products and Clean Technologies. July.
- Young, R. 1999. Personal communication with R. Young, Display Search, and M.L. Socolof, University of Tennessee, Center for Clean Products and Clean Technologies. August.
- VP150. 1998. Viewsonic VP150 17" monitor specifications sheet.
- Wood, S.P. 2000. Personal communication with S.P. Wood, Techneglas, and M.L. Socolof, University of Tennessee, Center for Clean Products and Clean Technologies. October 27.

Chapter 3

LIFE-CYCLE IMPACT ASSESSMENT

Within LCA, the LCI is a well established methodology; however, LCIA methods are less well defined and continue to evolve (Barnthouse *et al.*, 1997; Fava *et al.*, 1993). For toxicity impacts in particular, there are some methods being applied in practice (e.g., toxicity potentials, critical volume, and direct valuation) (Guinee *et al.*, 1996; ILSI, 1996; Curran, 1996), while others are in development. However, there is currently no general consensus among the LCA community as to one method over another. LCIA sophistication has also been discussed in efforts to determine the appropriate level of analytical sophistication for various types of decision making requirements (Bare *et al.*, 1999) or one that adequately addresses toxicity impacts.

Section 3.1 of this chapter presents the University of Tennessee (UT) LCIA methodology, which takes a more detailed approach to chemical toxicity impacts than some methods currently being used. Section 3.1 also discusses data sources, data quality, and the limitations and uncertainties in the LCIA methodology. The UT methodology calculates lifecycle impact category indicators for a number of impact categories, including several traditional LCA impact categories (e.g., global warming, stratospheric ozone depletion, photochemical smog, and energy consumption). Furthermore, the method calculates relative category indicators for potential chronic human health, aquatic ecotoxicity, and terrestrial ecotoxicity impacts in order to address interest in human and ecological toxicity and to fill a common gap in LCIAs. Work conducted for Saturn Corporation and the EPA Office of Research and Development by the UT Center for Clean Products and Clean Technologies has provided the basis for much of this methodology (Swanson, 2001).

Section 3.2 of this chapter describes the data management and analysis software used to calculate LCIA results. Section 3.3 presents the baseline LCIA results for both the CRT and the LCD. Baseline results are presented by impact category and include a discussion of the specific limitations and uncertainties in each category. Section 3.4 presents sensitivity analyses of the baseline results.

3.1 METHODOLOGY

In its simplest form, LCIA is the evaluation of potential impacts to any system as a result of some action. LCIAs generally classify the consumption and loading data from the inventory stage to various impact categories. Characterization methods are then used to quantify the magnitude of the contribution that loading or consumption could have in producing the associated impact. LCIA does not seek to determine actual impacts, but rather to link the data gathered from the LCI to impact categories and to quantify the relative magnitude of contribution to the impact category (Fava *et al.*, 1993; Barnthouse *et al.*, 1997). Further, impacts in different impact categories are generally calculated based on differing scales and therefore cannot be directly compared.

Conceptually, there are three major phases of LCIA, as defined by the Society of Environmental Toxicology and Chemistry (SETAC) (Fava *et al.*, 1993):

3.1 METHODOLOGY

- **Classification** The process of assignment and initial aggregation of data from inventory studies to impact categories (e.g., greenhouse gases or ozone depletion compounds).
- **Characterization** The analysis and estimation of the magnitudes of potential impacts for each impact category, derived through application of specific impact assessment tools. In the CDP, "impact scores" are calculated for inventory items that have been classified into various impact categories and then aggregated into life-cycle impact category indicators.
- **Valuation** The assignment of relative values or weights to different impacts and their integration across impact categories to allow decision makers to assimilate and consider the full range of relevant impact scores across impact categories.

The international standard for life cycle impact assessment, ISO 14042, considers classification and characterization to be mandatory elements of LCIA. Valuation or weighting is an optional element to be included depending on the goals and scope of the study. The CDP addresses the first two LCIA steps and leaves the valuation step to industry or others. In addition, further qualitative risk screening of selected materials is conducted beyond the traditional LCIA "characterization" phase in Chapter 4. The methodologies for life-cycle impact classification and characterization are described in Sections 3.1.1 and 3.1.2, respectively. Sections 3.1.3 and 3.1.4 address data sources and data quality, and limitations and uncertainties associated with the LCIA methodology.

3.1.1 Classification

In the first step of classification, impact categories of interest are identified in the scoping phase of the LCA. The categories to be included in the CDP LCIA are listed below:

- Natural Resource Impacts
 - renewable resource use
 - nonrenewable materials use/depletion
 - energy use
 - solid waste landfill use
 - hazardous waste landfill use
 - radioactive waste landfill use
- Abiotic Ecosystem Impacts
 - global warming
 - stratospheric ozone depletion
 - photochemical smog
 - acidification
 - air quality (particulate matter loading)
 - water eutrophication (nutrient enrichment)
 - water quality (biological oxygen demand [BOD] and total suspended solids [TSS])
 - radioactivity

- Potential Human Health and Ecotoxicity Impacts
 - chronic human health effects (occupational and public)
 - aesthetic impacts (odor)
 - aquatic ecotoxicity
 - terrestrial ecotoxicity

The second step of classification is assigning inventory inputs or outputs to applicable impact categories. Classification depends on whether the inventory item is an input or output, what the disposition of the output is, and in some cases the material properties for a particular inventory item. Figure 3-1 shows a conceptual model of classification for the CDP. Table 3-1 presents the inventory types and material properties used to define which impact category will be applicable to an inventory item. One inventory item may have multiple properties and therefore would have multiple impacts. For example, methane is both a global warming gas and has the potential to create photochemical oxidants (smog formation).

Output inventory items from a process may have varying dispositions, such as direct release (to air, water or land), treatment, or recycle/reuse. Outputs with direct release dispositions are classified into impact categories for which impacts will be calculated in the characterization phase of the LCIA. Outputs sent to treatment are considered inputs to a treatment process and impacts are not calculated until direct releases from that process occur. Similarly, outputs to recycle/reuse are considered inputs to previous processes and impacts are not directly calculated for outputs that go to recycle/reuse. Figure 3-1 graphically depicts the relationships between inventory type, dispositions, and impact categories. Note that a product is also an output of a process; however, product outputs are not used to calculate any impacts. Once impact categories for each inventory item are classified, life-cycle impact category indicators are quantitatively estimated through the characterization step.

3.1 METHODOLOGY



 * Equations for calculating impact scores for each impact category are provided in Sect. 3.1.2.

** Excluded from the scope of the CDP; however, included in the UT Life-Cycle Design Toolkit.

Note, radioactivity (not depicted in this figure) is classified for radioactive isotope outputs to air, water or landfill.

Figure 3-1. Impact classification conceptual model

Table 3-1.	Inventory types and properties for classifying inventory items
	into impact categories

Inventory Type		Chemical/Material Properties	Impact Category
Input	Output		
	•	Natural Resource Impacts	
material, water		renewable	renewable resource use
material, fuel		nonrenewable	nonrenewable resource use/depletion
electricity, fuel		energy	energy use
	solid waste to landfill	RCRA ^a - defined nonhazardous waste (or other country-specific definitions)	solid waste landfill use
	hazardous waste to landfill	RCRA ^a - defined hazardous waste (or other country-specific definitions)	hazardous waste landfill use
	radioactive waste to landfill	radioactive waste	radioactive waste landfill use
		Abiotic Ecosystem Impacts	
	air	global warming gases	global warming
	air	ozone depleting substances	stratospheric ozone depletion
	air	substances that can be photochemically oxidized	photochemical smog
	air	substances that react to form hydrogen ions (H+)	acidification
	air	air particulates (PM ₁₀ , TSP) ^a	air quality (air particulates)
	water	substances that contain available nitrogen or phosphorus	water eutrophication (nutrient enrichment)
	water	BOD ^a	water quality: BOD
	water	TSS ^a	water quality: TSS
	radioactivity to air, water, or land	radioactive substance (isotope)	radioactivity
Human Health and Ecotoxicity			
material		toxic material	chronic human health effects - occupational
	air, water	toxic material	chronic human health effects - public
	air	odorous material	aesthetic impacts (odor)
	water	toxic material	aquatic ecotoxicity
	air, water	toxic material	terrestrial ecotoxicity

^a Acronyms: Resource Conservation and Recovery Act (RCRA); particulate matter with average aerodynamic diameter less than 10 micrometers (PM₁₀); total suspended particulates (TSP); biological oxygen demand (BOD); total suspended solids (TSS).

3.1.2 Characterization

The characterization step of LCIA includes the conversion and aggregation of LCI results to common units within an impact category. Different assessment tools are used to quantify the magnitude of potential impacts, depending on the impact category. Three types of approaches are used in the characterization method for the CDP:

- **Loading** An impact score is based on the inventory amount.
- **Equivalency** An impact score is based on the inventory amount weighed by a certain effect, equivalent to a reference chemical.
 - *Full equivalency* all substances are addressed in a unified, technical model.
 - *Partial equivalency* a subset of substances can be converted into equivalency factors.
- **Scoring of inherent properties** An impact score is based on the inventory amount weighed by a score representing a certain effect for a specific material (e.g., toxicity impacts are weighed using a toxicity scoring method).

Table 3-2 lists the characterization approach used with each impact category. The loading approach either uses the direct inventory amount to represent the impact or slightly modifies the inventory amount to change the units into a meaningful loading estimate. Two examples are nonrenewable resource depletion and landfill use. Use of nonrenewable resources are directly estimated as the mass (loading) of that material consumed (input amount). Use of landfill space applies the mass loading of an output of hazardous, nonhazardous, or radioactive waste and converts that loading into a volume to estimate the amount of landfill space consumed.

The equivalency method uses equivalency factors that exist for certain impact categories. Equivalency factors are values that provide a relative measure or weighting that relate an inventory output amount to some impact category relative to a certain chemical. For example, to relate an atmospheric release to the global warming impact category, chemical-specific global warming potential (GWP) equivalency factors are used. GWPs are a measure of the possible warming effect on the earth's surface arising from the emission of a gas relative to carbon dioxide (CO_2). They are based on atmospheric lifetimes and radiative forcing of different greenhouse gases.

The scoring of inherent properties method is applied to impact categories that may have different effects for the same amount of various chemicals, but for which equivalency factors do not exist or are not widely accepted. The scores are meant to normalize the inventory data to provide measures of potential impacts. Scoring methods are employed for the human and ecological toxicity impact categories, based on the CHEMS-1 method described by Swanson et al. (1997), and presented below. The scoring method provides a hazard value (HV) for each potentially toxic material, which is then multiplied by the inventory amount to calculate the toxicity impact score. The aesthetics category directly applies an inherent chemical property (i.e., odor threshold concentration), but does not convert that value into a relative score, or HV.

Using the various approaches, the UT LCIA method calculates impact scores for each inventory item for each applicable impact category. Impact scores are therefore based on either a direct measure of the inventory amount or some modification (e.g., equivalency or scoring) of

that amount based on the potential effect the inventory item may have on a particular impact category. Impact scores are then aggregated within each impact category to calculate the various life-cycle impact category indicators.

Inventory amounts are identified on a functional unit basis and then used to calculate impact scores. For each inventory item, an individual score is calculated for each applicable impact category. The equations presented in the subsections that follow calculate impacts for individual inventory items that could later be aggregated as defined by the user. Impact scores represent relative and incremental changes rather than absolute effects or threshold levels.

Impact Category	Characterization Approach	
Natural Resource	e Impacts	
Renewable resource use	loading	
Nonrenewable materials use/depletion	loading	
Energy use	loading	
Solid waste landfill use	loading	
Hazardous waste landfill use	loading	
Radioactive waste landfill use	loading	
Abiotic Ecosystem	n Impacts	
Global warming	equivalency (full)	
Stratospheric ozone depletion	equivalency (full)	
Photochemical smog	equivalency (partial)	
Acidification	equivalency (full)	
Air quality (particulate matter)	loading	
Water eutrophication (nutrient enrichment)	equivalency (partial)	
Water quality (BOD, TSS)	loading	
Radioactivity	loading	
Human Health and Ecotoxicity		
Chronic human health effects - occupational	scoring of inherent properties	
Chronic human health effects - public	scoring of inherent properties	
Aesthetic impacts (odor)	application of inherent properties	
Aquatic ecotoxicity	scoring of inherent properties	
Terrestrial ecotoxicity	scoring of inherent properties	

Table 3-2. LCIA characterization approaches for the CDP

3.1 METHODOLOGY

3.1.2.1 Renewable and nonrenewable resource use

Natural resources are materials that are found in nature in their basic form rather than being manufactured (e.g., water, minerals, petroleum and wood). Renewable (or flow) resources, which are those that can be regenerated, are typically biotic resources (e.g., forest products, other plants or animals) and water. Nonrenewable (or stock) resources are abiotic, such as mineral ore or fossil fuels. Both of these natural resource impacts are calculated using the loading approach. Renewable and nonrenewable resource consumption impacts use direct consumption values (i.e., material mass) from the inventory.

Renewable resource impact scores are based on the following process inputs in the LCI: primary, ancillary, water, and fuel inputs of renewable materials. To calculate the loading-based impact scores, the following equation is used:

$$(IS_{RR})_i = [Amt_{RR} x (1 - RC)]_i$$

where:

 IS_{RR} equals the impact score for use of renewable resource i (kg) per functional unit; Amt_{RR} equals the inventory input amount of renewable resource i (kg) per functional
unit; andRCequals the fraction recycled content (post industrial and post consumer) of

resource *i*.

In the CDP LCI, most manufacturers that provided primary data did not report recycled content nor was the recycled content available for material inventories from secondary sources. Therefore, to calculate the impact score for use of renewable resources the recycled content (RC) was assumed to be zero.

Depletion of materials, which results from the extraction of renewable resources faster than they are renewed, may occur but is not specifically modeled or identified in the renewable resource impact score. For the nonrenewable materials use/depletion category, depletion of materials results from the extraction of nonrenewable resources. Nonrenewable resource impact scores are based on the amount of primary, ancillary, and fuel inputs of nonrenewable materials. To calculate the loading-based impact scores the following equation is used:

$$(IS_{NRR})_i = [Amt_{NRR} x (1 - RC)]_i$$

where:

 IS_{NRR} equals the impact score for use of nonrenewable resource *i* (NRR) (kg) per functional unit;

 Amt_{NRR} equals the inventory input amount of nonrenewable resource *i* (kg) per functional unit; and

RC equals the fraction recycled content (post industrial and post consumer) of resource *i*.

Due to the lack of data on the recycled content of nonrenewable resources, RC was assumed to be zero.

3.1.2.2 Energy use

General energy consumption is used as an indicator of potential environmental impacts from the entire energy generation cycle. Energy use impact scores are based on *fuel* and *electricity* inputs. The impact category indicator is the sum of electrical energy inputs and fuel energy inputs. Fuel inputs are converted from mass to energy units using the fuel's heat value (H) and the density (D), presented in Appendix K, Table K-1. The impact score is calculated by:

$$(IS_E)_i = Amt_{Ei} \text{ or } [Amt_F x (H/D)]_i$$

where:

$IS_E \\ Amt_E$	equals the impact score for energy use (MJ) per functional unit; equals the inventory input amount of electrical energy used (MJ) per functional unit;
Amt _F	equals the inventory input amount of fuel used (kg) per functional unit;
H	equals the heat value of fuel i (MJ/L); and
D	equals the density of fuel i (kg/L).

This category addresses energy *use* only. The emissions from energy production are outputs from the energy production process and are classified to applicable impact categories, depending on the disposition and chemical properties of the outputs (see Classification Section 3.1.1).

3.1.2.3 Landfill use

Landfill impacts are calculated using solid, hazardous, or radioactive waste outputs to land as volume of landfill space consumed. Solid waste landfill use pertains to the use of suitable and designated landfill space as a natural resource where municipal waste or construction debris is accepted. A solid waste landfill impact score is calculated using solid waste outputs disposed of in a solid waste (nonhazardous) landfill. Impact characterization is based on the volume of solid waste, determined from the inventory mass amount of waste and material density of each specific solid waste type:

$$(IS_{SWL})_i = (Amt_{SW}/D)_i$$

where:

 IS_{SWL} equals the impact score for solid waste landfill (SWL) use for waste i (m³) per
functional unit; Amt_{SW} equals the inventory output amount of solid waste i (kg) per functional unit; and
equals density of waste i (kg/m³).

Hazardous waste landfill use pertains to the use of suitable and designated landfill space as a natural resource where hazardous waste, as designated and regulated under the Resource Conservation and Recovery Act, is accepted. For non-US activities, equivalent hazardous or special waste landfills are considered for this impact category. Impact scores are characterized from hazardous waste outputs with a disposition of landfill. Impact characterization is based on the volume of hazardous waste, determined from the inventory mass amount of waste and material density of each specific hazardous waste type:

$$(IS_{HWL})_i = (Amt_{HW}/D)_i$$

where:

 IS_{HWL} equals the impact score for hazardous waste landfill (HWL) use for waste i (m³)
per functional unit; Amt_{HW} equals the inventory output amount of hazardous waste i (kg) per functional unit;

D equals density of waste i (kg/m³).

and

Radioactive waste pertains to the suitable and designated landfill space as a natural resource that accepts radioactive waste. Impacts are characterized from radioactive waste outputs with a disposition of landfill. Impact characterization is based on the volume of radioactive waste, determined from the inventory mass amount of waste and material density of each specific waste.

$$(IS_{RWL})_i = (Amt_{RW}/D)_i$$

where:

IS_{RWL}	equals the impact score for radioactive waste landfill (RWL) use for waste i (m ³)
	per functional unit;
Amt_{RW}	equals the inventory output amount of radioactive waste i (kg) per functional unit; and
D	equals density of waste i (kg/m ³).

3.1.2.4 Global warming impacts

The build up CO₂ and other greenhouse gases in the atmosphere may generate a "greenhouse effect" of rising temperature and climate change. Global warming potential (GWP) refers to the warming (relative to CO₂) that chemicals contribute to this effect by trapping the earth's heat. The impact scores for global warming (global climate change) effects are calculated using the mass of a global warming gas released to air modified by a GWP equivalency factor. The GWP equivalency factor is an estimate of a chemical's atmospheric lifetime and radiative forcing that may contribute to global climate change compared to the reference chemical CO₂. Therefore, GWPs are in units of CO₂ equivalents. GWPs have been published for known global warming chemicals within differing time horizons. The LCIA methodology being presented in this memorandum uses GWPs having effects in the 100-year time horizon. Although LCA does not necessarily have a temporal component of the inventory, these impacts are expected to be far enough into the future that releases occurring throughout the life cycle of a computer monitor would be within the 100-year time frame. Appendix K, Table K-2 presents a current list of GWPs as identified by the Intergovernmental Panel on Climate Change (IPCC) (Houghton et al., 1996). Global warming impact scores are calculated for any chemicals in the LCI that are found in Appendix K, Table K-2. The equation to calculate the impact score for an individual chemical is as follows:

$$(IS_{GW})_i = (EF_{GWP} \ x \ Amt_{GG})_i$$

where:	
IS_{GW}	equals the global warming impact score for greenhouse gas chemical i (kg CO ₂
	equivalents) per functional unit;
EF_{GWP}	equals the GWP equivalency factor for greenhouse gas chemical i (CO ₂
	equivalents, 100 year time horizon) (Appendix K, Table K-2); and
Amt_{GG}	equals the inventory output amount of greenhouse gas chemical <i>i</i> released to air
	(kg) per functional unit.

3.1.2.5 Stratospheric ozone depletion

The stratospheric ozone layer filters out harmful ultraviolet radiation from the sun. Chemicals such as chlorofluorocarbons, if released to the atmosphere, may result in ozonedestroying chemical reactions. Stratospheric ozone depletion refers to the release of chemicals that may contribute to this effect. Impact scores are based on the identity and amount of ozone depleting chemicals released to air. Currently identified ozone depleting chemicals are those with ozone depletion potentials (ODPs), which measure the change in the ozone column in the equilibrium state of a substance compared to the reference chemical chlorofluorocarbon (CFC)-11 (Heijungs *et al.*, 1992; CAAA, 1990). The list of ODPs that are used in this methodology are provided in Appendix K, Table K-3. The individual chemical impact score for stratospheric ozone depletion impacts is based on the ODP and inventory amount of the chemical:

$$(IS_{OD})_i = (EF_{ODP} \times Amt_{ODC})_i$$

where:

IS _{OD}	equals the ozone depletion impact score for chemical <i>i</i> (kg CFC-11 equivalents)
	per functional unit;
EF_{ODP}	equals the ODP equivalency factor for chemical <i>i</i> (CFC-11 equivalents)
	(Appendix K, Table K-3); and
Amt _{odc}	equals the amount of ozone depleting chemical <i>i</i> released to air (kg) per
	functional unit.

3.1.2.6 Photochemical smog

Photochemical oxidants are produced in the atmosphere from sunlight reacting with hydrocarbons and nitrogen oxides. At higher concentrations they may cause or aggravate health problems, plant toxicity, and deterioration of certain materials. Photochemical oxidant creation potential (POCP) refers to the release of chemicals that may contribute to this effect. The POCP is based on simulated trajectories of tropospheric ozone production with and without volatile organic carbons (VOCs) present. The POCP is a measure of a specific chemical compared to the reference chemical ethene (Heijungs *et al.*, 1992). The list of chemicals with POCPs to be used in this methodology are presented in Appendix K, Table K-4. As shown in Table 3-2, photochemical smog impacts are based on partial equivalency because some chemicals cannot be converted into POCP equivalency factors. For example, nitrogen oxides do not have a POCP. However, VOCs are assumed to be the limiting factor and if VOCs are present, there is a

potential impact. Impact scores are based on the identity and amount of chemicals with POCP equivalency factors released to the air and the chemical-specific equivalency factor:

 $(IS_{POCP})_i = (EF_{POCP} x Amt_{POC})_i$

IS _{POCP}	equals the photochemical smog impact score for chemical <i>i</i> (kg ethene
	equivalents) per functional unit;
EF_{POCP}	equals the POCP equivalency factor for chemical <i>i</i> (ethene equivalents)
	(Appendix K, Table K-4); and
Amt _{POC}	equals the amount of smog-creating chemical <i>i</i> released to the air (kg) per
	functional unit.

3.1.2.7 Acidification

This refers to the release of chemicals that may contribute to the formation of acid precipitation. Impact characterization is based on the amount of a chemical released to air that would cause acidification and the acidification potentials (AP) equivalency factor for that chemical. The AP equivalency factor is the number of hydrogen ions that can theoretically be formed per mass unit of the pollutant being released compared to sulfur dioxide (SO₂) (Heijungs *et al.*, 1992; Hauschild and Wenzel, 1997). Appendix K, Table K-5 lists the AP values that will be used as the basis of calculating acidification impacts. The impact score is calculated by:

$$(IS_{AP})_i = (EF_{AP} \times Amt_{AC})_i$$

where:

IS_{AP}	equals the impact score for acidification for chemical i (kg SO ₂ equivalents) per
	functional unit;
EF_{AP}	equals the AP equivalency factor for chemical i (SO ₂ equivalents) (Appendix K,
	Table K-5); and
Amt_{AC}	equals the amount of acidification chemical <i>i</i> released to the air (kg) per
	functional unit.

3.1.2.8 Air particulates

This refers to the release and build up of particulate matter primarily from combustion processes. Impact scores are based on particulate release amounts [particulate matter with average aerodynamic diameter less than 10 micrometers (PM_{10})] to the air. This size of particulate matter is most damaging to the respiratory system. Impact characterization is simply based on the inventory amount of particulates released to air. This loading impact score is calculated by:

$$IS_{PM} = Amt_{PM}$$

where:

 IS_{PM} equals impact score for particulates (kg PM₁₀) per functional unit, and

Amt_{PM} equals the inventory output amount of particulate release (PM₁₀) to the air (kg) per functional unit.

In this equation, PM_{10} is used to estimate impacts. However, if only total suspended particulates (TSP) data are available, these data may be used. Note that using TSP data is an overestimation of PM_{10} , which only refers to the fraction of particulates in the size range below 10 micrometers. A common conversion factor (TSP to PM_{10}) is not available because the fraction of PM_{10} varies depending on the type of particulates.

3.1.2.9 Water eutrophication

Eutrophication (nutrient enrichment) impacts to water are based on the identity and concentrations of eutrophication chemicals released to surface water after treatment. Equivalency factors for eutrophication have been developed assuming nitrogen (N) and phosphorus (P) are the two major limiting nutrients of importance to eutrophication. Therefore, the partial equivalencies are based on the ratio of N to P in the average composition of algae $(C_{106}H_{263}O_{110}N_{16}P)$ compared to the reference compound phosphate (PO_4^{3-}) (Heijungs *et al.*, 1992; Lindfors *et al.*, 1995). If the wastewater stream is first sent to a publicly owned treatment works (POTW), treatment is considered as a separate process and the impact score would be based on releases from the POTW to surface waters. Impact characterization is based on eutrophication potentials (EP) (Appendix K, Table K-6) and the inventory amount:

$$(IS_{EUTR})_i = (EF_{EP} x Amt_{EC})_i$$

where:

IS _{EUTR}	equals the impact score for regional water quality impacts from chemical i (kg
	phosphate equivalents) per functional unit;
EF_{EP}	equals the EP equivalency factor for chemical <i>i</i> (phosphate equivalents)
	(Appendix K, Table K-6); and
Amt_{EC}	equals the inventory output mass (kg) of chemical <i>i</i> per functional unit of
	eutrophication chemical in a wastewater stream released to surface water after
	any treatment, if applicable.

3.1.2.10 Water quality

Water quality impacts are characterized as surface water impacts due to releases of wastes causing oxygen depletion and increased turbidity. Two water quality impact scores are calculated based on the biological oxygen demand (BOD) and total suspended solids (TSS) in the wastewater streams released to surface water. The impact scores are based on releases to surface water following any treatment. Using a loading characterization approach, impact characterization is based on the amount of BOD and TSS in a wastewater stream. The water quality score equations for each are presented below:

 $(IS_{BOD})_i = (Amt_{BOD})_i$ and $(IS_{TSS})_i = (Amt_{TSS})_i$

where:	
IS _{BOD}	equals the impact score for BOD water quality impacts for waste stream i (kg) per
	functional unit;
Amt_{BOD}	equals the inventory output amount of BOD in wastewater stream <i>i</i> released to
	surface waters (kg) per functional unit;
IS _{TSS}	equals the impact score for TSS water quality impacts for waste stream i (kg) per
	functional unit; and
Amt_{TSS}	equals the inventory amount of TSS in wastewater stream <i>i</i> released to surface
	waters (kg) per functional unit.

3.1.2.11 Radioactivity

Radioactivity inventoried as the quantity of an isotope released to the environment is considered in the radioactivity impact category. These outputs, such as those from the generation of nuclear energy, can be air, water, or land releases. The radioactivity impact is a direct loading score measured in Bequerels of radioactivity, and calculated as follows:

$$(IS_{rad})_i = (Amt_{rad})_i$$

where:

 IS_{rad} equals the impact score for radioactivity of isotope *i* (Bq) per functional unit; and equals the inventory amount of radioactivity of isotope *i* (Bq) per functional unit.

While this impact category uses a loading approach, further refinement of this impact score calculation in the future could use radioactivity dose conversion factors, which convert radioactivity quantities (e.g., Bq) into human doses equivalents (e.g., sievert or rem).

3.1.2.12 Potential human health impacts

Human health impacts are defined in the context of life-cycle assessment as relative measures of potential adverse health effects to humans. Human health impact categories included in the scope of this LCA are chronic (repeated dose) effects, which include noncarcinogenic and carcinogenic effects, and aesthetics (although not a health effect *per se*, aesthetics pertains to human welfare). Chronic human health effects to both workers and the public are considered. Quantitative measures of consumer impacts are not included in this LCIA methodology because there are no direct outputs quantified in the LCI from the use stage of a computer monitor. The CDP does, however, quantify indirect outputs from energy consumption (i.e., pollutants released from energy production). In addition, Appendix L qualitatively discusses direct consumer impacts, such as electromagnetic radiation or eye strain.

The chemical characteristic that classifies inventory items to the human health effects (and ecotoxicity) categories is toxicity. Toxic chemicals were identified by searching lists of toxic chemicals [e.g., Toxic Release Inventory (TRI)] and if needed, toxicity databases [e.g., Hazardous Substances Data Bank (HSDB)], and Registry of Toxic Effects of Chemical Substances (RTECS), or other literature. Upon review by the EPA DfE Workgroup (see Appendix C), several materials in the CDP inventory were excluded from the toxic list if they are generally accepted as nontoxic. The EPA DfE Workgroup also reviewed the list of

chemicals that were included in this project as potentially toxic. The list of potentially toxic chemicals is provided in Appendix K, Table K-8, and chemicals that were excluded from the toxic list that appear in the CDP inventory are presented in Appendix K, Table K-9.

Human (and ecological) toxicity impact scores are calculated based on a chemical scoring method modified from CHEMS-1 found in Swanson *et al.* (1997). To calculate impact scores, chemical-specific inventory data are required. Any chemical that is assumed to be potentially toxic is given a toxicity impact score. If toxicity data are unavailable for a chemical, a mean default toxicity score is given. This is described in further detail below. Ecological toxicity is presented in Section 3.1.2.13.

Chronic human health effects are potential human health effects occurring from repeated exposure to toxic agents over a relatively long period of time (i.e., years). These effects could include carcinogenicity, reproductive toxicity, developmental effects, neurotoxicity, immunotoxicity, behavioral effects, sensitization, radiation effects, chronic effects to other specific organs or body systems (e.g., blood, cardiovascular, respiratory, kidney and liver effects). Impact categories for chronic health effects are divided into worker and public impacts. Occupational impact scores are based on inventory inputs and public impact scores are based on inventory outputs.

Chronic occupational health effects

This refers to potential health effects to workers, including cancer, from long-term repeated exposure to toxic or carcinogenic agents in an occupational setting. For possible occupational impacts, the identity and amounts of materials/constituents as input to a process are used. The inputs represent potential exposures and we could assume that a worker would continue to work at a facility and incur exposures over time. However, the inventory is based on manufacturing one monitor and does not truly represent chronic exposure. Therefore, the chronic health effects impact score is more a ranking of the potential of a chemical to cause chronic effects than a prediction of actual effects.

Chronic occupational health effects scores are based on the identity of toxic chemicals (or chemical ingredients) found in primary and ancillary inputs from materials processing and manufacturing life-cycle stages. The distinction between pure chemicals and mixtures is made implicitly, if possible, by specifying component ingredients of mixtures in the inventory.

The chronic human health impact scores are calculated using hazard values (HVs) for carcinogenic and for noncarcinogenic effects. The former HV uses cancer slope factors or cancer weight of evidence (WOE) classifications assigned by EPA and/or the International Agency for Research on Cancer (IARC) when no slope factor exists. If both an oral and inhalation slope factor exist, the slope factor representing the larger hazard is chosen. Where no slope factor is available for a chemical, but there is a WOE classification, the WOE is used to designate default hazard values as follows: EPA WOE Groups D (not classifiable) and E (noncarcinogen) and IARC Groups 3 (not classifications (known, probable, and possible human carcinogen) are given a default HV of 1 (representative of a mean slope factor) (Table 3-3). Similarly, materials for which no cancer data exist, but are designated as potentially toxic, are also given a default value of 1.

EPA Classification	IARC Classification	Description	Hazard Value	
Group A	Group 1	known human carcinogen	1	
Group B1	Group 2A	Probable human carcinogen (limited human data)	1	
Group B2	N/A	Probable human carcinogen (from animal data)	1	
Group C	Group 2B	Possible human carcinogen	1	
Group D	Group 3	Not classifiable	0	
Group E	Group 4	Noncarcinogenic or probably not carcinogenic	0	

Table 3-3. Hazard values for carcinogenicity weight-of-evidence if no slope factor is available

N/A: not applicable.

The cancer hazard value for chronic occupational health effects is the greater of the following:

oral:
$$(HV_{CA_{oral}})_i = \frac{oral SF_i}{oral SF_{mean}}$$

inhalation:
$$(HV_{CA_{inhalation}})_i = \frac{inhalation SF_i}{inhalation SF_{mean}}$$

where:

 HV_{CAoral} oral SF_i oral SF_{mean} equals the cancer oral hazard value for chemical *i* (unitless); equals the cancer oral slope factor for chemical *i* (mg/kg-day); equals the geometric mean cancer slope factor of all available slope factors (0.71 mg/kg-day); equals the cancer inhalation hazard value for chemical *i* (unitless); equals the cancer inhalation slope factor for chemical *i* (mg/kg-day)⁻¹; and equals the geometric mean cancer inhalation slope factor of all available inhalation slope factors (1.70 mg/kg-day)⁻¹.

The oral and inhalation slope factor mean values are the geometric means of a set of chemical data presented in Appendix K, Table K-10.

The noncarcinogen HV is based on either no-observed-adverse-effect levels (NOAELs) or lowest-observed-adverse-effect levels (LOAELs). The noncarcinogen HV is the greater of the

 $HV_{CAinhalation}$ inhalation SF_i inhalation SF_{mean}

oral:
$$(HV_{NC_{oral}})_i = \frac{1/(oral NOAEL_i)}{1/(oral NOAEL_{mean})}$$

inhalation and oral HV:

inhalation:
$$(HV_{NC_{inhalation}})_i = \frac{1/(inhal NOAEL_i)}{1/(inhal NOAEL_{mean})}$$

where:

$HV_{NC \ oral}$	equals the noncarcinogen oral hazard value for chemical <i>i</i> (unitless);
oralNOAEL _i	equals the oral NOAEL for chemical <i>i</i> (mg/kg-day);
oralNOAEL _{mean}	equals the geometric mean oral NOAEL of all available oral NOAELs
	(11.88 mg/kg-day);
HV _{NC inhalation}	equals the noncarcinogen inhalation hazard value for chemical <i>i</i> (unitless);
inhalNOAEL _i	equals the inhalation NOAEL for chemical $i \text{ (mg/m}^3)$; and
inhalNOAEL _{mean}	equals the geometric mean inhalation NOAEL of all available inhalation
	NOAELs (68.67 mg/kg-day).

The oral and inhalation NOAEL mean values are the geometric means of a set of chemical data presented in Appendix K, Table K-8. If LOAEL data are available instead of NOAEL data, the LOAEL divided by 10 is used to substitute for the NOAEL. The most sensitive endpoint is used if there are multiple data for one chemical.

The sum of the carcinogen and noncarcinogen HVs for a particular chemical is multiplied by the applicable inventory input to calculate the impact score:

$$(IS_{CHO})_i = [(HV_{CA} + HV_{NC}) x Amt_{TCinput}]_i$$

where:

IS _{CHO}	equals the impact score for chronic occupational health effects for chemical <i>i</i>
	(tox-kg) per functional unit;
HV_{CA}	equals the hazard value for carcinogenicity for chemical <i>i</i> ;
HV_{NC}	equals the hazard value for chronic noncancer effects for chemical <i>i</i> ; and
Amt TCinput	equals the amount of toxic inventory input (kg) per functional unit for chemical i.

Chronic public health effects

For chronic public health effects, the impact score represents a surrogate for potential health effects to residents living near a facility, including cancer, from long-term repeated exposure to toxic or carcinogenic agents. Impact scores are based on the identity and amount of toxic chemical outputs with dispositions to air and water.¹ As stated above, inventory items do not truly represent long-term exposure; instead, impacts are relative toxicity weightings of the inventory.

The scores for impacts to the public differ from the occupational impacts in that inventory outputs are used as opposed to inventory inputs. Note that this basic screening level scoring does not incorporate the fate and transport of the chemicals. The chronic public health effects impact score is calculated as follows:

$$(IS_{CHP})_{i} = [(HV_{CA} + HV_{NC}) \times Amt_{TCoutput}]_{i}$$

where:

equals the impact score for chronic human health effects to the public for IS_{CHP} chemical i (tox-kg) per functional unit; and equals the amount of toxic inventory output of chemical *i* to air and water (kg) per Amt_{TCoutput} functional unit.

Aesthetic impacts (odor)

This refers to impacts that detract from the quality of the local environment from a human perspective. Characterization in this project is based on odor. Impact scores are based on the identity and amount of odor-causing chemicals (Heijungs et al., 1992; EPA 1992), released to the air and their odor threshold value (OTV) (Heijungs et al., 1992) (Appendix K, Table K-7). This approach does not score chemicals as is done for toxic chemicals. The OTV is specific to a chemical, but does not use an equivalency factor that is based on a reference chemical or a hazard value based on a mean OTV. In this case, the OTV is a concentration which, when divided into the mass output of a chemical, results in an impact score in units of volume of malodorous air.

$$(IS_{AS})_i = (Amt_{OC}/OTV)_i$$

where:

IS_{AS}	equals the aesthetics impact score for chemical i (m ³ malodorous air) per functional unit
Amt _{oc}	equals the amount of odor-causing output for chemical <i>i</i> released to air (mg) per functional unit; and
OTV	equals the odor threshold value for chemical $i \text{ (mg/m}^3)$ (Appendix K, Table K-10).

¹ Disposition could be to groundwater. For example, a landfill model could have releases that go to groundwater.

Note that this impact assessment methodology determines the volume of malodorous air created if there is no dilution. In reality, many of the air releases reported in the LCI may occur at concentrations below the chemical's odor threshold.

3.1.2.13 Ecotoxicity

Ecotoxicity refers to effects of chemical outputs on nonhuman living organisms. Impact categories include ecotoxicity impacts to aquatic and terrestrial ecosystems.

Aquatic toxicity

Toxicity measures for fish are used to represent potential adverse effects to organisms living in the aquatic environment from exposure to a toxic chemical. Impact scores are based on the identity and amount of toxic chemicals as outputs to surface water. Impact characterization is based on CHEMS-1 acute and chronic hazard values for fish (Swanson *et al.*, 1997) combined with the inventory amount. Both acute and chronic impacts are combined into the aquatic toxicity term. The hazard values (HVs) for acute and chronic toxicity are based on LC₅₀ and NOAEL toxicity data, respectively, mostly from toxicity tests in fathead minnows (*Pimephales promelas*) (Swanson *et al.*, 1997). The acute fish HV is calculated by:

$$(HV_{FA})_i = \frac{1/(LC_{50})_i}{1/(LC_{50})_{mean}}$$

where:

 HV_{FA} equals the hazard value for acute fish toxicity for chemical *i* (unitless); LC_{50i} equals the lethal concentration to 50% of the exposed fish population for
chemical *i*; and LC_{50mean} equals the geometric mean LC_{50} of available fish LC_{50} values in Appendix K,
Table K-8 (23.45 mg/L).

The chronic fish HV is calculated by:

$$(HV_{FC})_i = \frac{1/NOAEL_i}{1/NOAEL_{mean}}$$

where:

 $\begin{array}{l} HV_{FC} \\ NOAEL_i \\ NOAEL_{mean} \end{array}$

equals the hazard value for chronic fish toxicity for chemical *i*; equals the no observed adverse affect level for fish for chemical *i*; and equals the geometric mean NOAEL of available fish NOAEL values in Appendix K, Table K-7 (3.90 mg/L). The aquatic toxicity impact score is calculated as follows:

$$(IS_{AQ})_i = [(HV_{FA} + HV_{FC}) x Amt_{TCoutput, water}]_i$$

where:

IS_{AQ}	equals the impact score for aquatic ecotoxicity for chemical i (tox-kg) per functional unit; and
$Amt_{TCoutput,water}$	equals the toxic inventory output amount of chemical <i>i</i> to water (kg) per functional unit.

Terrestrial toxicity

Toxicity measures for mammals (primarily rodents) are used to represent potential adverse effects to organisms living in the terrestrial environment from exposure to a toxic chemical. Impact scores are based on the identity and amount of toxic chemicals as outputs to air and surface water. Impact characterization is based on chronic toxicity hazard values combined with the inventory amount. The terrestrial toxicity impact score is based on the same noncancer chronic data used for human health because underlying data are from the same mammal studies (see Section 2.1.2.12 for the HV_{NC} term). The cancer hazard value was not included in the terrestrial impact score as it is based on ranking for potential human carcinogenicity. The terrestrial toxicity impact score is as follows:

$$(IS_{TER})_i = (HV_{NC} \ x \ Amt_{TCoutput})_i$$

where:

 IS_{TER} equals the impact score for terrestrial toxicity for chemical i (tox-kg) per
functional unit; and
equals the toxic inventory output amount of chemical i (kg) per functional unit.

3.1.2.14 Summary of impact score equations

Table 3-4 summarizes the impact categories, associated impact score equations, and the input or output data required for calculating natural resource impacts. Each of these characterization equations are loading estimates.

Impact Category Impact Score Approach		Data Required from Inventory (per functional unit)	
		Inputs	Outputs
Use of renewable resources	$IS_{RR} = Amt_{RR} \times (1 - RC)$	Material mass (kg) (e.g., water)	none
Use/depletion of nonrenewable materials	$IS_{NRR} = Amt_{NRR} x (1 - RC)$	Material mass (kg)	none
Energy use, general energy consumption	$IS_E = Amt_E \text{ or } (Amt_F x H/D)$	Energy (MJ) (electricity, fuel)	none
Solid waste landfill use	$IS_{SWL} = Amt_{SW} / D$	none	solid waste mass (kg) and density (i.e., volume, m ³)
Hazardous waste landfill use	$IS_{HWL} = Amt_{HW} / D$	none	hazardous waste mass (kg) and density (i.e., volume, m ³)
Radioactive waste landfill use	$IS_{RWL} = Amt_{RW} / D$	none	radioactive waste mass (kg) and density (i.e., volume, m ³)

Table 3-4. Summary of natural resources impact scoring

Abbreviations: RC = recycled content; H = heat value of fuel *i*; D = density of fuel *i*.

The term abiotic ecosystem refers to the nonliving environment that supports living systems. Table 3-5 presents the impact categories, impact score equations, and inventory data requirements for abiotic environmental impacts to atmospheric resources.

	Table 3-5.	Summary	of atmos	pheric resource	impact scoring
--	-------------------	---------	----------	-----------------	----------------

Impact Category	Impact Score Approach	Data Required from Inventory (per functional unit)	
		Inputs	Outputs
Global warming	$IS_{GW} = EF_{GWP} \times Amt_{GG}$	none	amount of each greenhouse gas chemical released to air
Stratospheric ozone depletion	$IS_{OD} = EF_{ODP} \times Amt_{ODC}$	none	amount of each ozone depleting chemical released to air
Photochemical smog	$IS_{POCP} = EF_{POCP} \times Amt_{POC}$	none	amount of each smog-creating chemical released to air
Acidification	$IS_{AP} = EF_{AP} \times Amt_{AC}$	none	amount of each acidification chemical released to air
Air quality (particulate matter)	$IS_{PM} = Amt_{PM}$	none	amount of particulates: PM_{10} or TSP released to air ^a

^a Assumes PM_{10} and TSP are equal; however, using TSP will overestimate PM_{10} .

Table 3-6 presents the impact categories, impact score equations, and required inventory data for abiotic environmental impacts to water resources.

Impact Category	Impact Score Approach	Data Required from Inventory (per functional unit)	
		Inputs	Outputs
Water eutrophication	$IS_{EUTR} = EF_{EP} \times Amt_{EC}$	none	amount of each eutrophication chemical released to water
Water quality (BOD)	$IS_{BOD} = Amt_{BOD}$	none	amount of BOD in each wastewater stream released to surface water
Water quality (TSS)	$IS_{TSS} = Amt_{TSS}$	none	amount of suspended solids (TSS) in each wastewater stream released to surface water

 Table 3-6.
 Summary of water resource impact scoring

Table 3-7 summarizes the human health and ecotoxicity impact scoring approaches. The impact categories, impact score equations, the type of inventory data, and the chemical properties required to calculate impact scores are presented. The human health effects and ecotoxicity impact scores are based on the scoring of inherent properties approach to characterization.

Impact Category	Impact Score Equations	Data Required from Inventory (per functional unit)		Chemical Properties Data Required
		Inputs	Outputs	
Chronic human health effects - occupational	$IS_{CHO} = (HV_{CA} + HV_{NC}) x$ $Amt_{TCinput}$	mass of each primary and ancillary toxic chemical	none	WOE or SF and/or mammal NOAEL or LOAEL
Chronic human health effects - public	$IS_{CHP} = (HV_{CA} + HV_{NC}) x$ $Amt_{TCoutput}$	none	mass of each toxic chemical released to air and surface water	WOE or SF and/or mammal NOAEL or LOAEL
Aesthetic impacts (odor)	$IS_{AS} = Amt_{OC} / OTV$	none	mass of odorous chemicals released to air	human odor threshold values
Aquatic toxicity	$IS_{AQ} = (HV_{FA} + HV_{FC}) x$ Amt _{TCoutput,water}	none	mass of each toxic chemical released to surface water	fish LC ₅₀ and/or fish NOAEL
Terrestrial toxicity	$IS_{TER} = HV_{NC} \times Amt_{TCoutput}$	none	mass of each toxic chemical released to air or surface water	mammal NOAEL

Table 3-7. Summary of human health and ecotoxicity impact scoring

3.1.2.15 Aggregation of impact scores

Individual impact scores are calculated for inventory items for a certain impact category and can be aggregated by inventory item (e.g., a certain chemical), process, life-cycle stage, or entire product profile. For example, global warming impacts can be calculated for one inventory item (e.g., CO_2 releases), for one process that could include contributions from several inventory items (e.g., electricity generation), for a life-cycle stage that may consist of several process steps (e.g., product manufacturing), or for an entire profile (e.g., a CRT desktop monitor over its life).

The following example illustrates how impacts are calculated. If two toxic chemicals [e.g., toluene and benzo(a)pyrene] are included in a waterborne release to surface water from Process A, impact scores would be calculated for the following impact categories (based on the classification shown in Table 3-1):

- Chronic public health effects;
- Aquatic toxicity; and
- Terrestrial toxicity.

Despite the output types being waterborne releases, the water eutrophication and water quality impact categories are not applicable here because the chemical properties criteria in Table 3-1 are not met. That is, these chemicals do not contain N or P and are not themselves wastewater streams.

Using chronic public health effects as an example, impact scores are then calculated for each chemical as follows:

IS _{CHP:toluene}	= $(HV_{CA:toluene} + HV_{NC:toluene}) \times Amt_{TCoutput:toluene}$
IS _{CHP:benzo(a)pyrene}	= $(HV_{CA:benzo(a)pyrene} + HV_{NC:benzo(a)pyrene}) x Amt_{TCoutput:benzo(a)pyrene}$

Table 3-8 presents toxicity data for the example chemicals from Appendix K, Table K-8. Using benzo(a)pyrene as an example, the hazard values are calculated as follows:

Table 5-6. Toxicity data used in example calculations				
Chemical	Cancer		Chronic noncancer effects	
	Weight of evidence	Slope factor (SF) (mg/kg-day) ⁻¹	Oral (mg/kg-day)	Inhalation (mg/m ³)
Toluene	D, 3	none	100 ^b	411.1 ^b
Benzo(a)pyrene	B2, 2A	3.1 ^a 7.3 ^c	no data	no data

Table 3-8. Toxicity data used in example calculations

^a inhalation SF

^b NOAEL

^c oral SF

Cancer effects:

oral:
$$(HV_{CA_{oral}})_i = \frac{oral SF_i}{oral SF_{mean}}$$

$$HV_{CAoral:benzo(a)pyrene} = 7.3 (mg/kg-day)^{-1} \div 0.71 (mg/kg-day)^{-1} = 10.3$$

inhalation:
$$(HV_{CA_{inh}})_i = \frac{inhalation SF_i}{inhalation SF_{mean}}$$

HV _{CAinhalation:benzo(a)pyrene}	$= 3.1 (mg/kg-day)^{-1} \div 1.7 (mg/kg-day)^{-1}$
	= 1.82

Thus, the cancer HV is 10.3, the greater of the two values.

Noncancer effects:

Since no data are available for noncancer effects, a default HV of one is assigned, representative of mean toxicity.

Total HV:

Thus the total hazard value for benzo(a)pyrene is given by:

$$HV_{benzo(a)pyrene} = HV_{CA} + HV_{NC}$$
$$= 10.3 + 1$$
$$= 11.3$$

Similarly, the HV for toluene is found to be 0.12. Given the following hypothetical output amounts:

 $Amt_{TC-O:TOLUENE} = 1.3 \text{ kg of toluene per functional unit}$ $Amt_{TC-O:BENZO(A)PYRENE} = 0.1 \text{ kg of benzo(a)pyrene per functional unit}$

the resulting impact scores are as follows:

$$\begin{split} \text{IS}_{\text{CHP-W:TOLUENE}} &= 0.12 \text{ x } 1.3 \\ \text{IS}_{\text{CHP-W:BENZO(A)PYRENE}} &= 11.3 \text{ x } 0.1 \\ &= 1.13 \text{ tox-kg of benzo(a)pyrene per functional unit} \end{split}$$
If these were the only outputs from Process A relevant to chronic public health effects, the total impact score for this impact category for Process A would be:

 $IS_{CHP:PROCESS_A} = IS_{CHP-W:TOLUENE} + IS_{CHP-W:BENZO(A)PYRENE}$ = 0.16 + 1.13 = 1.29 tox-kg per functional unit for Process A.

If the product system Y contained three processes altogether (Processes A, B, and C), and the impact scores for Process B and C were 2.5 and 3.0, respectively, impact scores would be added together to yield a total impact score for the product system relevant to chronic public health effects:

$$IS_{CHP:PROFILE_Y} = IS_{CHP:PROCESS_A} + IS_{CHP:PROCESS_B} + IS_{CHP:PROCESS_C}$$

= 1.29 + 2.5 + 3.0
= 6.8 tox-kg per functional unit for Profile Y.

An environmental profile would then be the sum of all the processes within that profile for each impact category.

3.1.3 Data Sources and Data Quality

Data that are used to calculate impacts are from: (1) equivalency factors or parameters used to identify hazard values; and (2) LCI items. Equivalency factors and data used to develop hazard values, which have been presented in this methodology, include GWP, ODP, POCP, AP, EP, WOE, SF, mammalian LOAEL/NOAEL, OTV, fish LC₅₀, and fish NOAEL. Published lists of the chemical-specific parameter values exist for GWP, ODP, POCP, AP, EP and OTV (see Appendix K). The other parameters may exist for a large number of chemicals and several data sources must be searched to identify the appropriate parameter values. Priority is given to peerreviewed databases (e.g., HEAST, IRIS, HSDB), then other databases (e.g., RTECS), other studies or literature, and finally estimation methods [e.g., structure-activity relationships (SARs) or quantitative structure-activity relationships (QSARs)]. The specific toxicity data that are used in the CDP are presented in Appendix K, Table K-8. The sources of each parameter presented in this report, and the basis for their values, are presented in Table 3-9. Data quality is affected by the type of data source (e.g., primary versus secondary data), the currency of the data, and the accuracy and precision of the data, and will depend on the source. The sources and quality of the LCI data used to calculate impact scores were discussed in Chapter 2. Data sources and data quality for each impact category are discussed further in Section 3.3, Baseline LCIA Results.

3.1.4 Limitations and Uncertainties

This section summarizes some of the limitations and uncertainties in LCIA methodology, in general. Specific limitations and uncertainties in each impact category are discussed in Section 3.3 with the baseline LCIA results.

The purpose of an LCIA is to evaluate the *relative potential* impacts of a product system for various impact categories. There is no intent to measure the *actual* impacts or provide spatial or temporal relationships linking the inventory to specific impacts. The LCIA is intended to

provide a screening-level evaluation of impacts. More detailed characterization of exposure and toxicity has been conducted on selected materials for the CDP in Chapter 4.

Parameter	Basis of Parameter Values	Source
Global warming potential (GWP)	atmospheric lifetimes and radiative forcing compared to CO_2	Houghton et al., 1996
Ozone depletion potential (ODP)	the change in the ozone column in the equilibrium state of a substance compared to CFC-11	Heijungs <i>et al.</i> , 1992; CAAA, 1990
Photochemical oxidant creation potential (POCP)	simulated trajectories of ozone production with and without VOCs present compared to ethene	Heijungs <i>et al.</i> , 1992
Acidification potential (AP)	number of hydrogen ions that can theoretically be formed per mass unit of the pollutant being released compared to SO_2	Heijungs <i>et al.</i> , 1992; Hauschild and Wenzel, 1997
Nutrient enrichment/eutrophication potential (EP)	ratio of N to P in the average composition of algae $(C_{106}H_{263}O_{110}N_{16}P)$ compared to phosphate (PO_4^{-3-})	Heijungs <i>et al.</i> , 1992; Lindfors <i>et al.</i> , 1995
Weight-of-evidence (WOE)	classification of carcinogenicity by EPA or IARC based on human and/or animal toxicity data	EPA, 1999; IARC, 1998
Slope factor (SF)	measure of an individual's excess risk or increased likelihood of developing cancer if exposed to a chemical, based on dose-response data	IRIS and HEAST as cited in Risk Assessment Information System (RAIS) online database
Mammalian: Lowest observed adverse effect level / No observed adverse effect level (LOAEL/NOAEL)	mammalian (primarily rodent) toxicity studies	IRIS, HEAST and various literature sources provided by EPA contractor
Fish lethal concentration to 50% of the exposed population (LC_{50})	fish (primarily fathead minnow) toxicity studies	Various literature sources and Ecotox database
Fish NOAEL	fish (primarily fathead minnow) toxicity studies	Literature sources and Ecotox database
Odor threshold value (OTV)	measured odor thresholds in humans	EPA, 1992

Fable 3-9.	Data sources	for e	quivalenc	y factors	and	hazard	values
------------	--------------	-------	-----------	-----------	-----	--------	--------

In addition to lacking temporal or spatial relationships and providing only relative impacts, LCA is also limited by the availability and quality of the inventory data. Data collection can be very time consuming and expensive. Confidentiality issues may also inhibit the availability of primary data.

Uncertainties are inherent in each parameter described in Table 3-9 and the reader is referred to each source for more information on associated uncertainties. For example, toxicity data require extrapolations from animals to humans and from high to low doses (for chronic effects) and can have a high degree of uncertainty.

Uncertainties also are inherent in chemical ranking and scoring systems, such as the scoring of inherent properties approach used for human health and ecotoxicity effects. In particular, systems that do not consider the fate and transport of chemicals in the environment can contribute to misclassifications of chemicals with respect to risk. Also, uncertainty is introduced where it was assumed that all chronic endpoints are equivalent, which is likely not the case. In addition, when LOAELs were not available but NOAELs were, a factor of ten was applied to the NOAEL to estimate the LOAEL, introducing uncertainty. The human health and ecotoxicity impact characterization methods presented here are screening tools that cannot substitute for more detailed risk characterization methods. However, it should be noted that in LCA, chemical toxicity is often not considered at all. This methodology is an attempt to consider chemical toxicity where it is often ignored.

Uncertainty in the inventory data depends on the responses to the data collection questionnaires and other limitations identified during inventory data collection. These uncertainties are carried into impact assessment. In this LCA, there was uncertainty in the inventory data, which included but was not limited to the following:

- missing individual inventory items,
- missing processes or sets of data,
- measurement uncertainty,
- estimation uncertainty,
- allocation uncertainty/working with aggregated data, and
- unspeciated chemical data.

The goal definition and scoping process helped reduce the uncertainty from missing data, although it is certain that some (missing data) still exist. As far as possible, the remaining uncertainties were reduced primarily through quality assurance/quality control measures (e.g., performing systematic double-checks of all calculations on manipulated data). The limitations and uncertainties in the inventory data were discussed further in Chapter 2.

3.2 DATA MANAGEMENT AND ANALYSIS SOFTWARE

The inventory and chemical characteristics data for the CDP are stored in a database within a software package developed by UT, using the Microsoft Visual FoxPro application programming language, under a cooperative agreement with the EPA Office of Research and Development. The software package calculates impact scores based on the stored inventory and chemical data and on the appropriate formulas for each impact category, as presented in Section 3.1.

This section presents the baseline LCIA results calculated using the impact assessment methodology presented in Section 3.1. As noted in the section on baseline LCI results (Section 2.7.1), the baseline scenario meets the following conditions:

- uses the effective life use stage scenario (e.g., use stage calculations are based on the actual amount of time a monitor is used by one or multiple users before it reaches its final disposition);
- uses the average value of all the energy inputs from the primary data for glass manufacturing;
- removes two outliers from the primary data for energy inputs during LCD panel/module manufacturing and then uses the average of the remaining energy inputs;
- excludes transportation in the manufacturing stage, but includes any transportation embedded in upstream data sets;
- includes the manufacturing processes of materials used as fuels (e.g., natural gas, fuel oil) in the manufacturing stage instead of in the upstream, materials processing stage. In cases where materials normally considered to be fuels are used as ancillary materials, their manufacturing processes are included with other upstream processes; and
- assumes LCD glass manufacturing processes use the same amounts of energy as CRT glass manufacturing per kilogram of glass produced.

Section 3.3.1 summarizes the baseline life cycle impact category indicators for both the CRT and LCD. Sections 3.3.2 through 3.3.14 present a breakdown of the impact category indicators by life-cycle stage, list the materials that contribute 99% of the total for both monitor types, and discuss limitations and uncertainties in each impact category. Each of the tables in this report shows the top contributors to the impacts because the complete tables, which are provided in Appendix J, are often lengthy. Section 3.3.15 summarizes the top contributors to each impact category, and Appendix M presents complete LCIA results.

3.3.1 Summary of Baseline LCIA Results

Table 3-10 presents the baseline CRT and LCD LCIA indicator results for each impact category, calculated using the impact assessment methodology presented in Section 3.1. The indicator results presented in the table are the result of the characterization step of LCIA methodology where LCI results are converted to common units and aggregated within an impact category. Note that the impact category indicator results are in a number of different units and therefore can not be summed or compared across impact categories. Note also that the CDP LCIA methodology does not perform the optional LCIA steps of normalization (calculating the magnitude of category indicator results relative to a reference value), grouping (sorting and possibly ranking of indicators), or weighting (converting and possibly aggregating indicator results across impact categories). Ranking and weighting, in particular, are subjective steps that depend on the values of the different individuals, organizations, or societies performing the analysis. Since the CDP involves a variety of stakeholders from different geographic regions and with different values, these more subjective steps were intentionally excluded from the CDP LCIA methodology.

Impact category	Units per monitor	CRT	LCD
Renewable resource use	kg	1.31E+04	2.80E+03
Nonrenewable resource use	kg	6.68E+02	3.64E+02
Energy use	MJ	2.08E+04	2.84E+03
Solid waste landfill use	m ³	1.67E-01	5.43E-02
Hazardous waste landfill use	m ³	1.68E-02	3.61E-03
Radioactive waste landfill use	m ³	1.81E-04	9.22E-05
Global warming	kg-CO ₂ equivalents	6.95E+02	5.93E+02
Ozone depletion	kg-CFC-11 equivalents	2.05E-05 ^{b,c}	1.37E-05 ^b
Photochemical smog	kg-ethene equivalents	1.71E-01	1.41E-01
Acidification	kg-SO ₂ equivalents	5.25E+00	2.96E+00
Air particulates	kg	3.01E-01	1.15E-01
Water eutrophication	kg-phosphate equivalents	4.82E-02	4.96E-02
Water quality, BOD	kg	1.95E-01	2.83E-02
Water quality, TSS	kg	8.74E-01	6.15E-02
Radioactivity	Bq	3.85E+07 ^d	1.22E+07 ^d
Chronic health effects, occupational	tox-kg	9.34E+02	6.96E+02
Chronic health effects, public	tox-kg	1.98E+03	9.02E+02
Aesthetics (odor)	m ³	7.58E+06	5.04E+06
Aquatic toxicity	tox-kg	2.25E-01	5.19E+00
Terrestrial toxicity	tox-kg	1.97E+03	8.94E+02

Table 3-10. Baseline life-cycle impact category indicators^a

^a Bold indicates the larger value within an impact category when comparing the CRT and LCD.

^b Several of the substances included in this category were phased out of production by January 1, 1996. Excluding phased out substances decreases the CRT ozone depletion indicator to 1.09E-05 kg CFC-11 equivalents per monitor and the LCD ozone depletion indicator to 1.18E-05 kg CFC-11 equivalents per monitor. These ozone depletion indicators are probably more representative of the CDP temporal boundaries and current operating practices. See Section 3.3.6 for details.

^c Although the CRT indicator appears larger than the LCD indicator, uncertainties in the inventory make it difficult to determine which monitor has the greater value. Therefore, this value is not shown in bold.

^d Radioactivity impacts are being driven by radioactive releases from nuclear fuel reprocessing in France, which are included in the electricity data in some of the upstream, materials processing data sets. See Section 3.3.12 for details.

As shown in the table, under the baseline conditions the CRT indicators are greater than the LCD indicators in the following categories: renewable resource use, nonrenewable resource use, energy use, solid waste landfill use, hazardous waste landfill use, radioactive waste landfill use, global warming, photochemical smog, acidification, air particulates, biological oxygen demand (BOD), total suspended solids (TSS), radioactivity, chronic public health effects, chronic occupational health effects, aesthetics, and terrestrial toxicity. The LCD indicators are greater than the CRT indicators in the following categories: water eutrophication and aquatic toxicity. In addition, as noted in Table 3-10, the CRT ozone depletion indicator is greater than that of the LCD when phased out substances are left in the CRT and LCD inventories. However, if phased out substances are removed from the CRT and LCD inventories, the LCD ozone depletion indicator would exceed that of the CRT.

A number of the impact results for both monitor types, and for the CRT in particular, are being driven by a few data points with relatively high uncertainty. Therefore, sensitivity analyses of the baseline results are presented in Section 3.4.

3.3.2 Renewable and Nonrenewable Resource Use

3.3.2.1 Renewable resource use

Figure 3-2 presents the CRT and LCD impact category indicators for renewable resource use by life-cycle stage, based on the impact assessment methodology presented in Section 3.1.2.1. Tables M-1 and M-2 in Appendix M present complete renewable resource results for the CRT and LCD, respectively. A renewable resource is one that is being replenished at a rate greater than or equal to its rate of depletion. Note that several of the resources listed in the Appendix and in the tables that follow are not renewable or can not be replenished, *per se*, but are considered renewable since they can be restored or are present in nearly infinite, nondepletable amounts. For example, water is typically considered a renewable resource since it can be restored to potable quality and is therefore being "replenished" at a rate greater than or equal to its rate of depletion. However, current trends toward shortages of potable water suggest that water might be more appropriately classified as a nonrenewable resource.



As shown in Figure 3-2, the baseline life-cycle impact category indicator for renewable resource use is 13,100 kg per monitor for the CRT and 2,800 kg per monitor for the LCD. Both the CRT and LCD renewable resource use results are dominated by the manufacturing life-cycle stage, with manufacturing accounting for 87% and 76% of the CRT and LCD totals, respectively.

Table 3-11 presents the life-cycle inventory items that contribute to the top 99% of the CRT renewable resource use total. It also lists the LCI data type (primary, secondary, or model/secondary). As shown in Table 3-11, water used in the production of LPG clearly dominates the CRT renewable resource use impact score. LPG is primarily used as an energy source in CRT glass manufacturing, indicating that the glass/frit process group is ultimately the greatest contributor to the CRT renewable resource use impact score. Other significant contributors include water used to produce electricity in the United States during the use of the monitor, water used in CRT tube manufacturing, and water used in the production of steel. The LCI data for LPG production and steel manufacturing are from secondary sources, while the LCI data for the U.S. electric grid are based on the model developed by the CDP for the amount of electricity consumed by a CRT during use combined with data from secondary sources on the inputs and outputs from U.S. power plants. CRT tube manufacturing LCI data are primary data collected by the CDP.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Manufacturing	LPG production	Water	Secondary	79%
Use	U.S. electric grid	Water	Model/secondary	8.7%
Manufacturing	CRT tube manufacturing	Water	Primary	6.2%
Materials processing	Steel production, cold-rolled, semi-finished	Water	Secondary	3.6%
Manufacturing	Japanese electric grid	Water	Model/secondary	0.34%
Manufacturing	PWB manufacturing	Water	Primary	0.32%

 Table 3-11. Top 99% of the CRT renewable resource use impact score

* Column may not add to 99% due to rounding.

Table 3-12 presents the inventory items contributing to the top 99% of the LCD renewable resource use total and the LCI data types (primary, secondary, or model/secondary). As shown in the table, water used in LCD module/monitor manufacturing is the greatest contributor to the LCD renewable resource use impact score. Other significant contributors include water used in the production of LPG, water used by the U.S. electric grid during the use life-cycle stage, and water used in steel production. It is LCD glass manufacturing that consumes the LPG responsible for the high LCD renewable resource use score. The LCI data for LCD module manufacturing are primary data collected by the CDP. LPG production and steel manufacturing are from secondary sources, while the LCI data for the U.S electric grid are based on the model developed by the CDP for the amount of electricity consumed by an LCD during the use stage combined with data from secondary sources.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Manufacturing	LCD module/monitor mfg.	Water	Primary	38%
Manufacturing	LPG production	Water	Secondary	18%
Use	U.S. electric grid	Water	Model/secondary	15%
Materials processing	Steel production (cold-rolled, semi-finished)	Water	Secondary	8.2%
Manufacturing	LCD panel components	Water	Primary	6.4%
Manufacturing	Backlight	Water	Primary	6.8%
Manufacturing	Japanese electric grid	Water	Model/secondary	5.3%
Manufacturing	PWB Manufacturing	Water	Primary	0.66%

 Table 3-12. Top 99% of the LCD renewable resource use impact score

* Column may not add to 99% due to rounding.

3.3.2.2 Nonrenewable resource use

Figure 3-3 presents the CRT and LCD impact category indicators for nonrenewable resource use by life-cycle stage, based on the impact assessment methodology presented in Section 3.1.2.1. Tables M-3 and M-4 in Appendix M present complete nonrenewable resource results for the CRT and LCD, respectively. The total nonrenewable resource use indicator was 668 kg per monitor for the CRT and 364 kg per monitor for the LCD. As shown in Figure 3-3, the CRT nonrenewable resource use results are dominated by the manufacturing life-cycle stage, which contributed 68% of the total. The LCD nonrenewable resource use score is dominated by the upstream materials processing stages, which contributed 69% of the total.



Table 3-13 presents the inventory items contributing to the top 99% of the CRT nonrenewable resource use impact score. It also lists the LCI data type (primary, secondary, or model/secondary). Similar to the renewable resource use LCIA results, the LPG production process, which mainly supports the CRT glass manufacturing process, clearly dominates the CRT nonrenewable resource use impact score. Petroleum used to make LPG is the non-renewable resource being consumed by the LPG production process in the greatest amounts, followed by natural gas, and coal. Note that the LPG actually consumed during CRT glass manufacturing does not appear in the nonrenewable resource use results. This is because it was accounted for in the nonrenewable resource use score for the LPG production process when it was extracted from the ground.

Fuels (coal and natural gas) consumed by the U.S. electric grid during monitor use are also among the greatest contributors to the CRT nonrenewable resource use impact scores. The LCI data for LPG production are from secondary sources, while the LCI data for the U.S. electric grid are based on the model developed by the CDP for the amount of electricity consumed by a CRT during use combined with data from secondary sources on the inputs and outputs from U.S. power plants.

Life-cycle stage	Process group	Material*	L CL data	Contribution
Life-cycle stage	i i occss gi oup	Waterial	tvne	to impact
			type	score*
Manufacturing	LPG production	Petroleum (in ground)	Secondary	56%
Use	U.S. electric grid	Coal, average (in ground)	Model/secondary	27%
Manufacturing	LPG production	Natural gas (in ground)	Secondary	6.7%
Use	U.S. electric grid	Natural gas	Model/secondary	2.1%
Manufacturing	LPG production	Coal, average (in ground)	Secondary	2.0%
Materials processing	Steel production, cold- rolled, semi-finished	Iron Ore	Secondary	0.99%
Materials processing	Steel production, cold- rolled, semi-finished	Coal, average (in ground)	Secondary	0.60%
Manufacturing	Fuel oil #6 production	Petroleum (in ground)	Secondary	0.58%
Use	U.S. electric grid	Petroleum (in ground)	Model/secondary	0.57%
Manufacturing	Natural gas production	Natural gas (in ground)	Secondary	0.51%
Manufacturing	U.S. electric grid	Coal, average (in ground)	Model/secondary	0.43%
Manufacturing	Japanese electric grid	Coal, average (in ground)	Model/secondary	0.34%
Materials processing	Aluminum production	Bauxite	Secondary	0.20%
Manufacturing	Japanese electric grid	Petroleum (in ground)	Model/secondary	0.19%
Materials processing	Polycarbonate production	Natural gas (in ground)	Secondary	0.19%
Manufacturing	Japanese electric grid	Natural gas	Model/secondary	0.19%

1 able 5-15. Top 99% of the CRT nonrenewable resource use impact sc	o 99% of the CRT nonrenewable resource use impact sco
---	---

* Column may not add to 99% due to rounding.

Table 3-14 presents the inventory items contributing to the top 99% of the LCD nonrenewable resource use impact score. In this case, the impact score is dominated by the natural gas extracted to produce natural gas in the upstream, materials processing life-cycle stage. Liquified natural gas (LNG) from this production process is used as an ancillary material in the LCD module/monitor manufacturing process group, indicating LCD module/monitor manufacturing is ultimately responsible for this non-renewable resource use. However, only one of the seven companies that provided data for the LCD module/monitor manufacturing process group reported this use of LNG. Note that the actual use of LNG in the LCD module/manufacturing process group does not appear in the nonrenewable resource results. Similar to the LPG results discussed above for the CRT, this is because it has been accounted for in the natural gas production process results.

Other primary contributors to this impact score include coal used to produce electricity for the U.S. electric grid, and petroleum used to produce LPG. The LCI data for all of the primary contributors to the LCD non-renewable resource use score were either from secondary sources or CDP models combined with secondary sources.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Materials processing	Natural gas production	Natural gas (in ground)	Secondary	65%
Use	U.S. electric grid	Coal, average (in ground)	Model/secondary	18%
Manufacturing	LPG production	Petroleum (in ground)	Secondary	4.9%
Manufacturing	Japanese electric grid	Coal, average (in ground)	Model/secondary	2.1%
Manufacturing	Natural gas production	Natural gas (in ground)	Secondary	1.5%
Use	U.S. electric grid	Natural gas	Model/secondary	1.4%
Manufacturing	Japanese electric grid	Petroleum (in ground)	Model/secondary	1.2%
Manufacturing	Japanese electric grid	Natural gas	Model/secondary	1.2%
Materials processing	Steel production (cold- rolled, semi-finished)	Iron ore	Secondary	0.89%
Manufacturing	LPG production	Natural gas (in ground)	Secondary	0.59%
Materials processing	Steel production (cold- rolled, semi-finished)	Coal (in ground)	Secondary	0.54%
Materials processing	Natural gas production	Coal (in ground)	Secondary	0.45%
Use	U.S. electric grid	Petroleum (in ground)	Model/secondary	0.39%

Table 3-14. Top 99% of the LCD nonrenewable resource use impact score

*Column may not add to 99% due to rounding

3.3.2.3 Limitations and uncertainties

The renewable and nonrenewable resource use results presented here are based on the mass of a material consumed. Depletion of renewable materials, which results from the extraction of renewable resources faster than they are renewed, may occur but is not specifically modeled or identified in the renewable resource impact scores. This may be particularly important for water, which, while considered a renewable resource, is in shorter and shorter supply as world population grows and more of the world's water resources become degraded. For the nonrenewable materials use category, depletion of materials results from the extraction

of nonrenewable resources. However, the impact scores do not directly relate consumption rates to the earth's ability to sustain that consumption.

The CRT and LCD impact scores for renewable resource use, and the CRT impact score for nonrenewable resources use, are being driven by the fuels consumed during CRT or LCD glass manufacturing. However, as discussed in Section 2.3.3.3, there is a high degree of variability in the three sets of CRT glass manufacturing energy data received by the CDP. Furthermore, as discussed in Section 2.3.3.1, LCD glass manufacturing data were developed from the CRT data because no companies were willing to supply the LCD data. Therefore, glass energy use inputs are uncertain for both the CRT and the LCD and were the subject of a sensitivity analysis, discussed in Section 3.4.

The LCD impact score for nonrenewable resource use is being driven by LNG used as an ancillary material during LCD module/monitor manufacturing. However, only one LCD module/monitor manufacturer reported using LNG as an ancillary material, which was confirmed by CDP researchers in follow-up communications. Given the fact that only one of seven manufacturers reported the ancillary use of LNG, the LCD nonrenewable resource use indicator may not be representative of the industry as a whole. If we remove this application of LNG from the LCD inventory, the LCD nonrenewable resource result is reduced by 66%, from 364 kg per monitor to 125 kg per monitor.

Inventory data for most of the materials contributing 99% of the CRT and LCD impact scores come from secondary sources, and were not developed specifically for the CDP. The limitations and uncertainties associated with secondary data sources are summarized in Section 2.2.2. Table 3-15 looks more closely at the LPG and natural gas production geographic and temporal boundaries. These are the production processes that are driving a large part of the CRT and LCD resources use indicators. As shown in the table, most of the LPG and natural gas production data are for the United States, although the LPG data set includes some data from other countries. Both data sets rely on several different sources and have different temporal boundaries. In particular, LPG production data are less recent, and may not accurately reflect current production practices. All of these factors create some inconsistencies among the data sets and reduce the data quality when used for the purposes of the CDP. However, this is a common difficulty with LCA, which often uses data from secondary sources to avoid the tremendous amount of time and resources required to collect all the needed data.

Production Process	Location	Source	Year
LPG production	Mainly U.S., but includes some other countries	Seven sources cited	1983 to 1993
Natural gas production	U.S.	Six sources cited	1987 to 1998

Table 3-15. LPG and natural gas production geographic and temporal boundaries

3.3.3 Energy Use

Figure 3-4 presents the CRT and LCD impact results for energy use by life-cycle stage, based on the impact assessment methodology presented in Section 3.1.2.2. Tables M-5 and M-6 in Appendix M list complete energy use results for the CRT and LCD, respectively. The total indicator for this impact category was 20,800 MJ per monitor for the CRT, and 2,840 MJ per monitor for the LCD.

CRTs generally are assumed to have greater life-cycle energy use impacts than the LCDs due to the high energy requirements in the use stage. This is borne out by the results in Figure 3-4, which show that CRT energy consumption during use is roughly 2.7 times that of the LCD. However, contrary to expectations, CRT energy use impacts are driven by the manufacturing life-cycle stage, which contributes about 88% of the total score. The use stage, which was expected to be responsible for a large amount of energy consumption impacts, only contributes about 11% of the total score. LCD energy consumption impacts are also largest in the manufacturing life-cycle stage which accounts for almost 51% of the impacts in this category. Both the use and upstream (materials processing) life-cycle stages are also significant contributors to LCD life-cycle energy use, accounting for 30 and 22%, respectively. Note that the sum of the upstream, manufacturing, and use life-cycle stages is greater than 100% due to an energy credit for incineration with energy recovery at the end of a monitor's useful life.



3.3.3.1 Major contributors to the CRT energy use results

Table 3-16 presents the life-cycle inventory items contributing to the top 99% of the CRT energy use results and the LCI data type (primary, secondary, or model/secondary). As shown in the table, LPG used in the glass/frit process group, primarily from CRT glass manufacturing, clearly dominates the CRT energy use result, followed by electricity consumed during use of a CRT monitor, and natural gas, petroleum, and coal consumed during LPG production. Since LPG is used primarily as an energy source during CRT glass manufacturing, most of the sum of the glass/frit manufacturing and LPG production energy use impacts—roughly 87% of the CRT life-cycle energy use impacts—can be attributed to the CRT glass manufacturing process.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Manufacturing	Glass/frit manufacturing	Liquified petroleum gas	Primary	72%
Use	CRT monitor use	Electricity	Model/secondary	11%
Manufacturing	LPG production	Natural gas (in ground)	Secondary	10%
Manufacturing	LPG production	Petroleum (in ground)	Secondary	2.0%
Manufacturing	LPG production	Coal, average (in ground)	Secondary	1.4%
Manufacturing	CRT tube manufacturing	Fuel oil #6	Primary	0.72%
Manufacturing	Glass/frit manufacturing	Natural gas	Primary	0.26%
Manufacturing	Glass/frit manufacturing	Fuel oil # 2	Primary	0.24%
Manufacturing	Glass/frit manufacturing	Electricity	Primary	0.23%
Manufacturing	CRT tube manufacturing	Natural gas	Primary	0.18%
Materials processing	Polycarbonate production	Natural gas (in ground)	Secondary	0.16%
Manufacturing	LPG production	Uranium (U, ore)	Secondary	0.16%
Manufacturing	CRT tube manufacturing	Electricity	Primary	0.15%
Manufacturing	Glass/frit manufacturing	Electricity	Primary	0.13%

 Table 3-16.
 Top 99% of the CRT energy use impact score

*Column may not add to 99% due to rounding.

3.3.3.2 Major contributors to the LCD energy use results

Table 3-17 lists the inventory items contributing to the top 99% of the LCD life-cycle energy use results. Electricity consumed during use of the LCD monitor by the consumer is the single largest contributor to LCD energy use impacts, closely followed by LPG utilized to produce LCD glass. Other major contributors include natural gas consumed during natural gas production, electricity and LNG used as a fuel during LCD monitor/module manufacturing, and natural gas consumed during LPG production. Note that the LNG used as an ancillary material in LCD module/monitor manufacturing is not included in the LCD energy use impact calculations since it is not used as a source of energy. However, natural gas used as an energy source to produce the LNG is included (the third item listed in the table).

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Use	LCD monitor use	Electricity	Model/ secondary	30%
Manufacturing	LCD glass manufacturing	Liquified petroleum gas	Primary	25%
Materials processing	Natural gas production	Natural gas (in ground)	Secondary	14%
Manufacturing	LCD module/monitor mfg.	Electricity	Primary	8.9%
Manufacturing	LCD module/monitor mfg.	Liquified natural gas	Primary	5.8%
Manufacturing	LPG production	Natural gas (in ground)	Secondary	3.4%
Manufacturing	LCD panel components	Electricity	Primary	1.4%
Manufacturing	LCD module/monitor mfg.	Natural gas	Primary	1.3%
Materials processing	Natural gas production	Coal (in ground)	Secondary	1.3%
Materials processing	Natural gas production	Petroleum (in ground)	Secondary	0.89%
Manufacturing	LCD module/monitor mfg.	Liquified petroleum gas	Primary	0.88%
Manufacturing	LPG production	Petroleum (in ground)	Secondary	0.69%
Materials processing	Polycarbonate production	Natural gas (in ground)	Secondary	0.67%
Manufacturing	LPG production	Coal (in ground)	Secondary	0.51%
Manufacturing	LCD module/monitor mfg.	Kerosene	Primary	0.46%
Materials processing	PMMA sheet production	Petroleum (in ground)	Secondary	0.42%
Materials processing	PMMA sheet production	Natural gas (in ground)	Secondary	0.41%
Materials processing	Steel production (cold-rolled, semi-finished)	Petroleum (in ground)	Secondary	0.36%
Materials processing	Steel production (cold-rolled, semi-finished)	Natural gas (in ground)	Secondary	0.35%
Manufacturing	Natural gas production	Natural gas (in ground)	Secondary	0.32%
Materials processing	PMMA sheet production	Electricity	Secondary	0.32%
Materials processing	Steel production (cold-rolled, semi-finished)	Electricity	Secondary	0.31%
Manufacturing	LCD module/monitor mfg.	Fuel oil # 4	Primary	0.31%
Materials processing	Styrene-butadiene copolymer production	Natural gas (in ground)	Secondary	0.29%
Materials processing	Aluminum production	Coal (in ground)	Secondary	0.29%

Tuble e 11. Top >> /o of the heb cherg, use impact score
--

*Column may not add to 99% due to rounding.

3.3.3.3 Limitations and uncertainties

Some of the limitations and uncertainties in the energy use indicators are similar to those in the renewable and nonrenewable resource use categories. First, as discussed in Section 3.3.1.1, the energy data for both CRT and LCD glass manufacturing are uncertain due to the variability in the primary glass data received by the CDP from three glass manufacturers. Glass manufacturing energy data are the subject of a sensitivity analysis in Section 3.4. Second, data for LPG production and natural gas production, which are among the largest contributors to the energy use indicators, are from secondary sources and are therefore subject to the limitations and

uncertainties associated with secondary data (see Section 2.2.2 and 3.3.1.1). Not counting the use stage data, note that about 14% of the CRT energy use impacts shown in Table 3-16, above, are from secondary sources, compared to about 24% of the LCD energy use impacts (Table 3-17).

The amount of electricity consumed during use of a monitor was modeled by the CDP from secondary sources on the amount of electricity consumed during different power modes and the amount of time a monitor spends in each mode. Data quality for the effective life scenario (the baseline scenario presented here) is considered to be excellent, based on the source and quality information detailed in Appendix H and discussed in Sections 2.4.2 and 2.4.3.

3.3.4 Landfill Use

3.3.4.1 Solid waste landfill use

Figure 3-5 presents the CRT and LCD LCIA results for the solid waste landfill use impact category, based on the impact assessment methodology presented in Section 3.1.2.3. Tables

M-7 and M-8 in Appendix M present complete results for the CRT and LCD, respectively. Lifecycle solid waste landfill use was 0.17 m³ for the CRT and 0.054 m³ for the LCD. The solid waste landfill indicators for both monitor types are dominated by waste disposal during the use stage—which contributes 59% of the total for the CRT and 68% for the LCD—primarily from wastes generated as a by-product of electricity production. Both monitor types have negative solid waste impact scores during the end-of-life stage. This is due to an energy credit from incineration processes, which offsets some of the solid waste impacts from electricity generation.



Table 3-18 presents the materials that contribute to the top 99% of the CRT solid waste landfill use impact score. Note that the material contributions actually add to greater than 100% due to the energy credit from incineration processes, discussed above. Coal waste from U.S. electricity production is the single largest contributor to CRT impacts in this impact category, followed by slag and ash from LPG production, and dust/sludge and fly bottom ash from U.S.

electricity production. Electricity is used to power the monitor during the use stage, while LPG is primarily used in the production of CRT glass during manufacturing. LPG production also results in an unspecified solid waste that contributes 4.5% of the CRT solid waste impact score, while the CRT glass/frit process group generates a wastewater treatment sludge that contributes 5% of the score. Thus, the CRT glass/frit process group contributes about 30% of the CRT solid waste impact score, either directly (as a result of the glass manufacturing process itself) or indirectly (from LPG production). Other processes that are significant contributors include steel production and the landfilling of a CRT at the end of its effective life. The latter value is based on the assumption that 25% of CRTs that have reached their end of life are disposed of in a solid waste landfill (see Section 2.5 and Appendix I).

Life-cycle stage	Process group	Material LCI data type		Contribution to impact score*
Use	U.S. electric grid	Coal waste	Model/secondary	38%
Manufacturing	LPG production	Slag and ash	Secondary	21%
Use	U.S. electric grid	Dust/sludge	Model/secondary	12%
Use	U.S. electric grid	Fly bottom ash	Model/secondary	10%
Materials processing	Steel production, cold- rolled, semi-finished	Unspecified solid waste	Secondary	6.6%
End-of-Life	CRT landfilling	EOL CRT monitor, landfilled	Primary	5.0%
Manufacturing	LPG production	Unspecified waste	Secondary	4.5%
Manufacturing	CRT glass/frit mfg.	Waste water treatment sludge	Primary	4.4%

Table 3-18. Top 99% of the CRT solid waste landfill use impact score

*Column adds to greater than 100% due to a credit from incineration with energy recovery during the EOL life-cycle stage.

CRT glass manufacturing data were collected specifically for the CDP, while data for other process groups were either modeled by the CDP from secondary sources (e.g., U.S. electric grid data) or are entirely from secondary sources (e.g., LPG and steel production data). The mass and volume of CRT materials that are landfilled were developed for the CDP based on the mass reported in each inventory data set (collected as primary data) and the density of CRT materials assumed to be disposed of in a solid waste landfill. Note that the upstream inventories, which were derived from secondary sources (i.e., *Ecobilan*), include electricity generation within the materials manufacturing processes. These inventories do not include coal waste as an output, but list "slag and ash" as an output. The different inventories used in this project have varying nomenclature and some of the solid waste materials listed in the table may indeed overlap.

Table 3-19 presents the materials that contribute to the top 99% of the LCD solid waste landfill use results. Like the CRT solid waste results, the material contributions in the table are actually greater than 100% due to some negative values at end-of-life from an incinerator energy credit. Coal waste, dust/sludge, and fly/bottom ash from U.S. electricity during the use stage dominate the LCD solid waste impacts, contributing 68% of the impact score. Other significant contributors include the following: (1) an unspecified solid waste from producing steel used in the manufacture of the monitor, (2) slag and ash generated during the production of natural gas which is then used by one LCD module/monitor manufacturer as an ancillary material (LNG) during LCD module/monitor manufacture, (3) a wastewater treatment sludge from LCD

module/monitor manufacturing, (4) coal waste from the generation of electricity in Japan during manufacturing, and (5) landfilling of non-hazardous or non-recovered components of the LCD at the end of its effective life. The latter value is based on the assumption that 50% of LCDs are sent to a solid waste landfill at the end of their effective lives.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Use	U.S. electric grid	Coal waste	Model/secondary	44%
Use	U.S. electric grid	Dust/sludge	Model/secondary	13%
Use	U.S. electric grid	Fly/bottom ash	Model/secondary	11%
Materials processing	Steel production, cold- rolled, semi-finished	Unspecified solid waste	Secondary	9.9%
Materials processing	Natural gas production	Slag and ash	Secondary	7.7%
Manufacturing	LCD monitor/module mfg.	Waste water treatment sludge	Primary	5.6%
Manufacturing	Japanese electric grid	Coal waste	Model/secondary	5.0%
End-of-Life	LCD landfilling	EOL LCD monitor, landfilled	Primary	3.5%

 Table 3-19. Top 99% of the LCD solid waste landfill use impact score

*Column adds to greater than 100% due to a credit from incineration with energy recovery during the EOL life-cycle stage.

LCI data for LCD monitor/module manufacturing were collected by the CDP, and LCD solid waste disposal volumes were estimated by the CDP based on the amounts and density of LCD materials assumed to be disposed of in a solid waste landfill. Like the CRT, data for other process groups either were modeled by the CDP from secondary data sources or came from secondary sources. As discussed above for the CRT, the materials processing inventories from secondary sources (i.e., *Ecobilan*) include electricity generation within the materials manufacturing processes. The different inventories used in this project have varying nomenclature and some of the solid waste materials listed in the table may indeed overlap.

3.3.4.2 Hazardous waste landfill use

Figure 3-6 presents the CRT and LCD LCIA results for the hazardous waste landfill use impact category, based on the impact assessment methodology presented in Section 3.1.2.3. Tables M-9 and M-10 (see Appendix M) present complete hazardous waste landfill use results for the CRT and LCD, respectively. Hazardous waste landfill use impacts are characterized from hazardous waste outputs with a disposition of landfill, which includes about 83% of the 9.46 kg of hazardous waste/functional unit generated by the CRT life cycle and about 27% of the 6.3 kg/functional unit generated by the LCD life cycle. This consumes approximately 0.017 m³ of hazardous waste landfill space for the CRT and 0.036 m³ for the LCD, based on the mass and densities of the various materials. The results for both monitor types are dominated by monitor disposal at the end of its effective life. Approximately 46% of CRTs and 5% of LCDs are assumed to be landfilled as hazardous waste (see Section 2.5 and Appendix I).



Table 3-20 presents the materials that contribute to the top 99% of the CRT hazardous waste landfill use results. About 91% of the total hazardous waste landfill space consumed throughout the life-cycle of the CRT is from the amount of the monitor that is assumed to be disposed of as hazardous waste. The next largest contributor is an unspecified hazardous waste from LPG production. Most of this LPG is used to manufacture CRT glass. CRT outputs to a hazardous waste landfill at the end-of-life were estimated by the CDP. The LPG inventory is from secondary sources.

Table 3-20. Top 99% of the CRT hazardous waste landfill use impact score

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score
End-of-Life	CRT landfilling	EOL CRT monitor, landfilled	Primary	91%
Manufacturing	LPG production	Hazardous waste	Secondary	8.1%

Table 3-21 lists the top contributors to the LCD hazardous waste landfill use results. LCD results are also dominated by landfilling of the LCD monitor at the end of its effective life, even though only 5% of LCDs are assumed to be landfilled. Other significant contributors include an unspecified hazardous waste from LPG production, and acetic acid from LCD monitor/module manufacturing. LPG is used in the manufacture of LCD glass. LCD outputs to a hazardous waste landfill at the end-of-life were estimated by the CDP. The LPG inventory is from secondary sources.

Table 3-21. Top 99% of the LCD hazardous waste landfill use impact score

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score
End-of-Life	LCD landfilling	EOL LCD monitor, landfilled	Primary	97%
Manufacturing	LPG production	Hazardous waste	Secondary	1.8%
Manufacturing	LCD monitor/module mfg.	Acetic acid	Primary	0.88%

3.3.4.3 Radioactive waste landfill use

Figure 3-7 presents the CRT and LCD LCIA results for the radioactive waste landfill use impact category, based on the impact assessment methodology presented in Section 3.1.2.3. Tables M-11 and M-12 in Appendix M present complete results for the CRT and LCD, respectively. Life-cycle radioactive waste landfill use indicators for the CRT are 1.81E-04 m³ per monitor for the CRT and 9.22E-05 m³ per monitor for the LCD. As shown in the figure, CRT radioactive waste landfill impacts are dominated by radioactive waste disposal in the use stage, which contributes about 79% of the total impacts. This result is to be expected, given the relatively large amount of electricity consumed by a CRT during use and the associated radioactive waste from nuclear power plants. The use stage also contributes the greatest amount of LCD impacts (58%), but the manufacturing stage is also a significant contributor (33%) due to electricity consumed during manufacturing. LCD manufacturing electricity is linked to the Japanese electric grid, which derives 31% of its power from nuclear sources. By comparison, about 20% of the U.S. electric grid is powered by nuclear sources.



Table 3-22 lists the materials that contribute to the top 99% of the CRT radioactive waste landfill use score and the LCI data type. Note that the LCI data for all of the materials in the table are from secondary sources or models. Low-level radioactive waste and depleted uranium from the U.S. electric grid are the radioactive materials being landfilled in the greatest quantities, followed by low-level radioactive waste from the production of steel used in the monitor. The latter radioactive waste is a byproduct of electricity production used in the manufacture of steel. It should be noted that the electricity generation data utilized in the steel inventory are from France (Glazebrook, 2001), where a large percentage of electricity is derived from nuclear sources. Therefore, these emissions may not be representative of emissions from steel production in some parts of Asia or in the United States. This issue is discussed further in the section on limitations and uncertainties, below.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Use	U.S. electric grid	Low-level radioactive waste	Model/secondary	61%
Use	U.S. electric grid	Uranium, depleted	Model/secondary	18%
Materials processing	Steel production, cold- rolled, semi-finished	Low-level radioactive waste	Secondary	9.0%
Manufacturing	Japanese electric grid	Low-level radioactive waste	Model/secondary	3.8%
Materials processing	Invar	Low-level radioactive waste	Secondary	2.6%
Materials processing	Ferrite manufacturing	Low-level radioactive waste	Secondary	2.5%
Manufacturing	Japanese electric grid	Uranium, depleted	Model/secondary	1.1%
Manufacturing	U.S. electric grid	Low-level radioactive waste	Model/secondary	0.97%

Table 3-22	. Top 99% of the	CRT radioactive waste	e landfill use impact score
-------------------	------------------	------------------------------	-----------------------------

*Column may not add to 99% due to rounding.

Other significant contributors to the CRT radioactive waste score include low-level radioactive waste from electricity used in Japan during manufacturing, and low-level radioactive waste from invar and ferrite manufacturing. Invar is an alloy of nickel and iron. Like the steel data discussed above, the invar and ferrite manufacturing data also include emissions from electricity production. Finally, low-level radioactive waste from U.S. electricity consumed during the manufacturing stage contributes slightly less than 1% of the CRT radioactive waste landfill use impacts. The frit and PWB manufacturing processes consume this electricity. These are the only CRT components linked to the U.S. grid.

Table 3-23 lists the materials that contribute to the top 99% of the LCD radioactive waste landfill use results and the LCI data type. Note that LCI data for all of the primary contributors to this impact category are from secondary sources or modeled from secondary sources. Together, low-level radioactive waste and depleted uranium disposal from electricity consumed during use of the monitor account for about 57% of the LCD radioactive waste landfill use indicator, followed by low-level radioactive waste and depleted uranium disposal from electricity used during manufacturing (roughly 32%). Waste disposal from steel production is also a significant contributor at 8.7%. Like the CRT data discussed above, these emissions occur from electricity production in France and may not be representative of U.S. or some Asian practices.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Use	U.S. electric grid	Low-level radioactive waste	Model/secondary	44%
Manufacturing	Japanese electric grid	Low-level radioactive waste	Model/secondary	25%
Use	U.S. electric grid	Uranium, depleted	Model/secondary	13%
Materials processing	Steel production, cold- rolled, semi-finished	Low-level radioactive waste	Secondary	8.7%
Manufacturing	Japanese electric grid	Uranium, depleted	Model/secondary	7.5%

Table 3-23. Top 99% of the LCD radioactive waste disposal imp	pact score
---	------------

*Column may not add to 99% due to rounding.

3.3.4.4 Limitations and uncertainties

Landfill use pertains to the use of suitable and designated landfill space as a natural resource where the specified type of waste (solid, hazardous, or radioactive) is accepted. Landfill use impacts are characterized from solid, hazardous, or radioactive waste outputs with a disposition of landfill. Impact characterization is based on the volume of waste determined from the inventory mass amount of waste and materials density of each specific waste. Note that different countries may have different landfill designations for the final disposition of similar waste streams (e.g., a waste considered hazardous in the U.S. may be accepted in a solid waste landfill elsewhere). However, where possible, equivalent landfills (e.g., special waste landfills and hazardous waste landfills) were considered for these impact categories.

CRT and LCD impact results for the solid and radioactive waste landfill use categories were driven almost entirely by waste outputs reported in inventories from secondary sources. These inventories were not developed specifically for the CDP and therefore are subject to the limitations and uncertainties associated with secondary data (see Section 2.2.2 and 3.3.2). In particular, radioactive waste disposal from some of the upstream materials processing data may not be representative of conditions in the U.S. or parts of Asia. These data include emissions from electricity production within the materials inventory, which are the primary source of radioactive waste streams. For example, steel production data include the French electric grid, where a large percentage of the power supply comes from nuclear power plants. In addition, some of the upstream data may not be representative of current conditions, with steel production data covering the period from 1975 to 1990 and invar production data being from 1991.

CRT and LCD impact results for the hazardous waste landfill use category were dominated by monitor disposal at the end of their effective lives. Hazardous waste landfill disposal volumes were estimated based on the percent of monitors with hazardous waste landfilling as their final disposition, the monitor mass, and the material densities. However, data on the percentage of CRTs that are landfilled are not separated into hazardous and nonhazardous landfilling processes. Therefore, these percentages were estimated by the CDP, as described in Appendix I. Even less is known about the final disposition of LCDs, particulary since very few LCD desktop monitors have reached the end of their effective lives (and then, only if they have been damaged in some way). Therefore, the effect of different LCD EOL dispositions was evaluated in a sensitivity analysis (see Section 3.4.)

3.3.5 Global Warming

Figure 3-8 presents the CRT and LCD LCIA results for the global warming impact category, based on the impact assessment methodology presented in Section 3.1. Tables M-13 and M-14 in Appendix M list complete global warming results for the CRT and LCD, respectively. The life-cycle global warming indicators for the CRT and LCD were 695 and 593 kg of CO_2 equivalents per monitor, respectively. The CRT global warming indicators are driven by the use life-cycle stage, which contributes about 66% of the total. The manufacturing stage, which contributed 88% of the CRT energy consumption impacts, only contributes about 29% of the total global warming score. LCD global warming impacts, on the other hand, have the greatest contribution from the manufacturing life-cycle stage, which accounts for about 40% of the potential impacts in this category. Both the upstream (materials processing) and use

life-cycle stages are also significant contributors to the LCD global warming results, accounting for 31% and 29% of the total, respectively.



One might expect the distribution of global warming impacts across life-cycle stages to mirror those of energy consumption, as CO_2 is generally a large emission from electricity generation. However, as discussed in Section 3.3.3, CRT energy impacts are greatest in the manufacturing stage due to the large amounts of energy used to manufacture glass. Since the energy used in glass manufacturing is not only from electricity, but more so from other fuels (LPG, natural gas, and fuel oil), there is not a direct correlation between CRT global warming impacts and CRT energy impacts.

The distribution of LCD global warming impacts across life-cycle stages does mirror the distribution of LCD energy use impacts discussed in Section 3.3.3. However, as discussed below, the manufacturing stage global warming impacts for the LCD are being driven more by the use of sulfur hexafluoride (SF₆) in LCD monitor/module manufacturing than by the use of electricity.

3.3.5.1 Major contributors to the CRT global warming results

Table 3-24 presents the life-cycle inventory items contributing to the top 99% of the CRT global warming results and the LCI data type (primary, secondary, or model/secondary). As shown in the table, CRT global warming impacts are dominated by CO_2 emissions from electricity generation during use of the monitor, followed by CO_2 and methane emissions from producing LPG used as fuel in the CRT glass/frit process group. Together these three emissions contribute almost 89% of the CRT life-cycle global warming score. Carbon dioxide and methane emissions from a number of other processes also add to the CRT global warming score, as does nitrous oxide emissions from the LPG production process. It is likely that most of the CO_2 emissions from the materials processing life-cycle stage can be attributed to emissions from electricity generation or fuel combustion. As discussed in Section 2.2.2.1, the upstream materials processing inventories used in this study include data from electricity generation.

Note that almost all of the LCI data for global warming emissions are from secondary sources. This is because the CRT global warming results are dominated by CO₂ emissions from electricity generation, and electric grid data were either developed by the CDP from secondary sources or already included in the upstream, materials processing inventories from secondary sources.

	_			~
Life-cycle stage	Process group	Material	LCI data	Contribution
			type	to impact
				score*
Use	U.S. electric grid	Carbon dioxide	Model/secondary	64%
Manufacturing	LPG production	Carbon dioxide	Secondary	22%
Manufacturing	LPG production	Methane	Secondary	2.5%
Manufacturing	Japanese electric grid	Carbon dioxide	Model/secondary	2.2%
Use	U.S. electric grid	Methane	Model/secondary	1.9%
Materials processing	Steel production, cold-rolled,	Carbon dioxide	Secondary	1.9%
	semi-finished			
Manufacturing	U.S. electric grid	Carbon dioxide	Model/secondary	1.0%
Manufacturing	LPG production	Nitrous oxide	Secondary	0.72%
Materials processing	Polycarbonate production	Carbon dioxide	Secondary	0.66%
Materials processing	Aluminum production	Carbon dioxide	Secondary	0.52%
End-of-Life	CRT incineration	Carbon dioxide	Secondary	0.51%
Manufacturing	CRT glass/frit mfg.	Carbon dioxide	Primary	0.40%
Materials processing	Invar	Carbon dioxide	Secondary	0.33%
Materials processing	Styrene-butadiene copolymer	Carbon dioxide	Secondary	0.24%
	production			

Table 3-24. Top 99% of the CRT global warming impact score

*Column may not add to 99% due to rounding.

3.3.5.2 Major contributors to the LCD global warming results

Table 3-25 presents the life-cycle inventory items contributing to the top 99% of the LCD global warming results and the LCI data type (primary, secondary, or model/secondary). Sulfur hexafluoride used in LCD module manufacturing is the single largest contributor to LCD global warming impacts, followed by CO_2 emissions from electricity generation during the use stage, CO_2 and methane emissions from natural gas production, and CO_2 emissions from the generation of electricity used during manufacturing in Japan.

Life-cycle stage	Process group	Material	LCI data	Contribution to
			type	impact score*
Manufacturing	LCD monitor/module mfg.	Sulfur hexafluoride	Primary	29%
Use	U.S. electric grid	Carbon dioxide	Model/secondary	28%
Materials processing	Natural gas production	Carbon dioxide	Secondary	16%
Materials processing	Natural gas production	Methane	Secondary	12%
Manufacturing	Japanese electric grid	Carbon dioxide	Model/secondary	8.7%
Manufacturing	LPG production	Carbon dioxide	Secondary	1.2%
Materials processing	Steel production, cold- rolled, semi-finished	Carbon dioxide	Secondary	1.1%
Use	U.S. electric grid	Methane	Model/secondary	0.85%
Materials processing	PMMA sheet production	Carbon dioxide	Secondary	0.45%
Materials processing	Polycarbonate production	Carbon dioxide	Secondary	0.43%
Manufacturing	Natural gas production	Carbon dioxide	Secondary	0.35%
End-of-Life	LCD incineration	Carbon dioxide	Secondary	0.35%

 Table 3-25.
 Top 99% of the LCD global warming impact score

*Column may not add to 99% due to rounding.

Sulfur hexafluoride is a potent global warming gas, with a global warming potential (GWP) equivalancy factor of 23,900 CO₂ equivalents (see Table K-2 in Appendix K). It is used as an etchant in a dry-etching process of amorphus silicon and SiNx films. The CO₂ and methane emissions from natural gas production can be attributed to the use of LNG as an ancillary material in LCD monitor/module manufacturing. However, as discussed in the section on non-renewable resource use (Section 3.3.2), only one LCD module manufacturer reported this use of LNG.

Carbon dioxide emissions (and, in one case, methane emissions) round out the remainder of the primary contributors to the LCD global warming indicator. Most of the carbon dioxide emissions occur from upstream processes and are due to electricity generation. With the exception of the SF₆ data, the LCI data for all of the top LCD global warming emissions are from secondary sources. Sulfur hexaflouride emissions data were developed by the CDP based on an emissions factor (0.45) applied to SF₆ inputs reported by LCD monitor/module manufacturers. The emissions factor is from the Intergovernmental Panel on Climate Change publication, *Good Practice Guidance and Uncertainty Management in National Greenhouse Gas Inventories* (Penman *et al.*, 2000.)

3.3.5.3 Limitations and uncertainties

Global warming potential (GWP) refers to the warming that emissions of certain gases may contribute by building up in the atmosphere and trapping the earth's heat. As discussed in Section 3.1.2.4, the LCIA methodology for global warming impacts uses published GWP equivalency factors having effects in the 100-year time horizon. These effects are expected to be far enough into the future that releases occurring throughout the life cycle of a computer monitor would be within the 100-year timeframe.

The effects of the buildup of global warming gases in the atmosphere is still the subject of scientific debate, but in 1995 the Intergovernmental Panel on Climate Change (IPCC), representing the consensus of most climate scientists worldwide, concluded that "... the balance of evidence...suggests that there is a discernible human influence on global climate (IPCC, 1995)." Other than the limitations and uncertainties inherent in predicting future effects, most of the limitations and uncertainties in the CRT and LCD global warming results have to do with the LCI data on greenhouse gas emissions, which occur primarily from electricity generation processes.

As noted above, the U.S. and Japan electric grid inventories used in the CDP were developed by the CDP, and electric grids used with upstream processes are embedded in the upstream inventories. U.S. electric grid emissions of CO_2 are based on data in the EPA publication, *National Air Quality and Emissions Trends Report, 1997* (EPA, 1998), which were the best data available when the electric grid inventory data was developed and are expected to be reasonably accurate. However, the Japanese electric grid inventory was derived from the U.S. inventory based on the mix of fuels used in Japan. Because Japanese power plants may employ different pollution control devices, use fuels of different quality, or other factors, their greenhouse gas emissions could actually be higher or lower than those reported in the inventory.

Similarly, the electric grid inventories embedded in upstream, materials processing inventories may have differing geographic and temporal boundaries or may be representative of older technologies. Therefore, actual emissions of greenhouse gases could be higher or lower than reported. This is a common limitation of LCAs, which must often rely on secondary data sources to avoid the considerable time and resources required to collect primary data for every process.

3.3.6 Stratospheric Ozone Depletion

Figure 3-9 presents the CRT and LCD LCIA results for the stratospheric ozone depletion impact category by life-cycle stage, based on the impact assessment methodology presented in Section 3.1.2.5. Note that most of the CRT ozone depletion impacts occur from the use stage (50%) and the upstream, materials processing stages (45%), while most of the LCD ozone depletion impacts occur from the manufacturing stage (63%). As will be shown later, this is important because upstream and use stage data are primarily from secondary data sources, whereas manufacturing data were collected by the CDP. Tables M-15 and M-16 in Appendix M list complete stratospheric ozone depletion results for the CRT and LCD, respectively.



The ozone depletion impact category indicator was 2.05E-05 kg of CFC-11 equivalents per monitor for the CRT and 1.37E-05 kg of CFC-11 equivalents per monitor for the LCD. However, for both the CRT and the LCD, many of the materials contributing to this impact category are listed as Class I ozone depleting substances in Title VI of the 1990 Clean Air Act Amendments, and therefore were required to be phased out of U.S. production by January 1, 1996. Production of these substances was also phased out in other developed countries under the Montreal Protocol and its Amendments and Adjustments, but continues today in some developing countries. An exception is bromomethane, which is a Class I substance that will not be completely phased out of production until 2005 (EPA, 2001a).

For a few of the phased out substances, a significant amount of inventory remained after production was phased out. However, most of these inventories are now exhausted, and Class I ozone depleting substances are rarely used by manufacturers in developed countries. If we delete the phased out substances from the CRT and LCD inventories, the CRT ozone depletion indicator is reduced 47% from 2.05E-05 to 1.09E-05 kg of CFC-11 equivalents per monitor, and the LCD result is reduced 14% from 1.37E-05 to 1.18E-05 kg of CFC-11 equivalents per monitor. These latter values are probably more representative of the temporal boundaries for primary data collected in the CDP LCA. Thus, when all data are included in the ozone depletion calculations, the CRT has a greater ozone depletion impact score than the LCD, but the results are switched (LCD greater than CRT) when phased out substances are removed from the inventory.

3.3.6.1 Major contributors to the CRT ozone depletion results

Table 3-26 lists the materials that contribute to the top 99% of the CRT life-cycle ozone depletion impact score and the LCI data type. Bromomethane emissions from electricity generated in the use stage are the single largest contributor to the CRT ozone depletion indicator, accounting for almost half of the total score. Most of the other materials in the table are emitted from materials production processes in the upstream, materials processing life-cycle stage. Exceptions are bromomethane emissions from the LPG production process (used to produce

LPG for the glass/frit process group), 1,1,1-trichloroethane emissions from electricity generation in the use stage, and bromomethane emissions from electricity used in manufacturing.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score ^a
Use	U.S. electric grid	Bromomethane ^b	Model/secondary	49%
Materials processing	ABS production	HALON-1301 ^b	Secondary	20%
Materials processing	Aluminum production	HALON-1301 ^b	Secondary	14%
Materials processing	Invar	HALON-1301 ^b	Secondary	5.9%
Manufacturing	LPG production	Bromomethane ^b	Secondary	3.7%
Materials processing	Lead	HALON-1301 ^b	Secondary	2.6%
Materials processing	Steel production, cold- rolled, semi-finished	HALON-1301 ^b	Secondary	2.0%
Use	U.S. electric grid	1,1,1-Trichloroethane ^b	Model/secondary	1.1%
Manufacturing	U.S. electric grid	Bromomethane ^b	Model/secondary	0.78%

 Table 3-26. Top 99% of the CRT stratospheric ozone depletion impact score

^a Column may not add to 99% due to rounding.

^b Class I substance as listed in Title VI of the Clean Air Act Amendments.

Note that all of the materials listed in Table 3-26 are Class I ozone depleting substances. As discussed above, all of these substances except bromomethane were phased out of production by January 1, 1996. Note also that all of the LCI data for the materials in the table are from secondary sources. For both of these reasons, the LCI data for the materials in the table are highly uncertain. This is discussed further under limitations and uncertainties, below.

3.3.6.2 Major contributors to the LCD ozone depletion results

Table 3-27 lists the top contributors to the LCD life-cycle stratospheric ozone depletion indicator and the LCI data type. Together HCFC-225cb and HCFC 225ca used in the LCD panel components process group account for 59% of the LCD ozone depletion indicator. Note that HCFC 225cb and HCFC 225ca are Class II ozone depleting substances that are not scheduled for phaseout until 2015. [Under U.S. regulations and the Montreal Protocol and its Amendments and Adjustments these substances can not be produced or imported after 2015, except for use as refrigerants in equipment manufactured before January 1, 2020 (EPA, 2001b).] Also note that the impact scores for these materials are based on primary LCI data collected from manufacturers. Therefore, these data are considered to be more reliable than data for Phase I substances from secondary sources.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score ^a
Manufacturing	LCD panel components	HCFC-225cb ^b	Primary	34%
Use	U.S. electric grid	Bromomethane ^c	Model/secondary	27%
Manufacturing	LCD panel components	HCFC-225ca ^b	Primary	25%
Materials processing	Aluminum production (virgin)	HALON-1301°	Secondary	7.8%
Manufacturing	Japanese electric grid	Bromomethane ^c	Model/secondary	3.1%
Materials processing	Steel production, cold- rolled, semi-finished	HALON-1301°	Secondary	1.4%

Table 3-27. Top 99% of the LCD stratospheric o	ozone depletion use impact score
--	----------------------------------

^a Column may not add to 99% due to rounding.

^b Class II substance as listed in Title VI of the Clean Air Act Amendments.

° Class I substance as listed in Title VI of the Clean Air Act Amendments.

Other significant contributors to the LCD ozone depletion score include bromomethane emissions from electricity generation in the United States and Japan, and halon emissions from upstream, materials processing stages. As noted above in the discussion of CRT ozone depletion results, the halons were phased out of production in 1996, which suggests that these data are not representative of current conditions. Bromomethane is still being produced, but bromomethane emissions data are also uncertain as will be discussed in the section on limitations and uncertainties, below.

3.3.6.3 Limitations and uncertainties

Both the CRT and LCD life-cycle stratospheric ozone depletion results are highly uncertain due to the inclusion of a number of Class I ozone depleting substances in inventories from secondary sources. As discussed above, except for bromomethane, developed countries that are parties to the Montreal Protocol and its Amendments and Adjustments phased out the production of Class I substances by 1996. To better assess the uncertainties in these results, Table 3-28 lists the geographic and temporal boundaries for the life-cycle inventories of the process groups listed in tables 3-26 and 3-27, above.

Process group Geographic boundari		Temporal boundaries		
ABS production	Germany, Italy, Netherlands	1997		
Aluminum production	Not provided	Not provided		
Invar production	Multiple countries	1991 (nickel), Not provided (lead)		
Japanese electric grid	U.S. and Japan ^a	1993 ^b		
LCD panel components	Japan	1998		
Lead production	Not provided	Not provided		
LPG production	Mainly U.S.	1983-1993		
Steel production	Multiple countries	1975-1990		
U.S. electric grid	U.S.	1993 ^b		

 Table 3-28. Geographic and temporal boundaries of inventories contributing to the CRT and/or LCD ozone depletion indicator results

^a Based on the U.S. electric grid inventory modified to account for the fuel mix used in Japan.

^b Date of stack tests from which bromomethane emission factor was developed (from EPA Web site: Emission Factor Documentation for AP-42 Section 1.1: Bituminous and Subbituminous Coal Combustion. http://www.epa.gov/ttn/chief/ap42/ch01/bgdocs/b01s01.pdf.)

The most recent data are for LCD panel components manufacturing, which are primary data collected from manufacturers in Japan by the CDP and expected to be of better quality than older data from secondary sources. Data for ABS production are also fairly recent, dating from 1997. However, the temporal boundaries for most of the data are either not listed in the inventories, or pre-date the Class I substance production phase out. In addition, most of the data are from Europe and/or the United States, where very few Class I ozone depleting substances are currently used. Thus, we suspect that emissions of Class I substances reported in the inventories no longer occur, indicating that the CRT and LCD life-cycle impact results should be reduced to 2.05E-05 kg of CFC-11 equivalents per monitor for the CRT, and 1.37E-05 kg of CFC-11 equivalents per monitor for the CRT.

Bromomethane is a Class I ozone depleting substance that has not yet been phased out of production and is emitted during coal combustion to produce electricity. Bromoethane emissions from electricity production are estimated from an emission factor reported in AP-42, the EPA compilation of air pollutant emission factors (EPA, 1996). EPA (1996) provides an emission factor rating that is, "an overall assessment of how good a factor is, based on both the quality of the test(s) or information that is the source of the factor and on how well the factor represents the emission source." The bromomethane emissions factor rating is "D," or below average, indicating CDP data quality for bromomethane emissions from electricity generation is also below average.

In conclusion, it appears that most of the Class I substance emissions data are highly uncertain or of below average quality. Manufacturing data collected by the CDP, which includes emissions of Class II substances, are of better quality and expected to be more representative of current conditions. When all data are included in the ozone depletion calculations, the CRT has a greater ozone depletion impact score than the LCD. However, if we remove phased out substances from the inventory, the results are switched with the LCD having a greater score in this category than the CRT.

3.3.7 Photochemical Smog

Figure 3-10 presents the CRT and LCD LCIA results for the photochemical smog impact category by life-cycle stage. These results were calculated using the impact assessment methodology presented in Section 3.1.2.6. Tables M-17 and M-18 in Appendix M list complete results for the CRT and LCD, respectively. One 17" CRT monitor produces 0.171 kg of ethene equivalents throughout its life cycle, while a functionally equivalent 15" LCD monitor produces 0.141 of ethene equivalents. The CRT photochemical smog impact score is dominated by emissions during the manufacturing stage (71% of total); the LCD impact score is dominated by emissions during the upstream, materials processing stages (91% of total). However, as discussed below, it is fossil fuel production processes that emit the majority of smog forming emissions during the life-cycle of either monitor type. Both the CRT and LCD receive a slight credit on emissions of smog forming chemicals at the end of their effective lives due to energy recovery from incineration processes.



3.3.7.1 Major contributors to the CRT photochemical smog results

Table 3-29 lists the materials that contribute to the top 99% of the CRT photochemical smog indicator result. The LPG production process alone, which emits various unspeciated hydrocarbons, benzene, aldehydes, ethane, and formaldehyde, accounts for almost 67% of CRT photochemical smog impacts. As noted earlier in the discussion of other impact category indicators, most of this LPG is used as a fuel source in CRT glass manufacturing. However, CRT glass energy data reported in the three data sets received by the CDP were highly variable and therefore the subject of a sensitivity analysis (see Section 3.4).

Other materials responsible for more than 1% of the CRT photochemical smog score include the following: (1) hydrocarbon (methane and nonmethane) emissions associated with steel production, (2) methane emissions from electricity generation during the use stage, (3) toluene emissions from CRT tube manufacturing, (4) nonmethane hydrocarbon emissions associated with ABS production, and (5) nonmethane hydrocarbon emissions associated with polycarbonate production. Note that the inventory for each upstream material (e.g., steel

production, ABS production, etc.) contains data from the raw materials extraction, materials manufacture, and (usually) electricity generation processes. Therefore, hydrocarbon emissions associated with steel production, for example, could be from one of many individual processes, such as the steel production process itself, from fuels consumed during the mining of ore, or from the combustion of fuels as an energy source during steel manufacturing.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact
			·5 P ·	score*
Manufacturing	LPG production	Hydrocarbons, unspeciated	Secondary	36%
Manufacturing	LPG production	Nonmethane hydrocarbons, unspeciated	Secondary	25%
Materials processing	Steel production, cold- rolled, semi-finished	Nonmethane hydrocarbons, unspeciated	Secondary	19%
Manufacturing	LPG production	Methane	Secondary	3.4%
Use	U.S. electric grid	Methane	Model/secondary	2.6%
Materials processing	Steel production, cold- rolled, semi-finished	Hydrocarbons, unspeciated	Secondary	2.0%
Manufacturing	LPG production	Benzene	Secondary	1.6%
Manufacturing	CRT tube mfg.	Toluene	Primary	1.3%
Materials processing	ABS production	Nonmethane hydrocarbons, unspeciated	Secondary	1.3%
Materials processing	Polycarbonate production	Nonmethane hydrocarbons, unspeciated	Secondary	1.1%
Materials processing	Aluminum production	Nonmethane hydrocarbons, unspeciated	Secondary	0.86%
Manufacturing	Natural gas production	Nonmethane hydrocarbons, unspeciated	Secondary	0.53%
Materials processing	ABS production	Nonmethane hydrocarbons, unspeciated	Secondary	0.41%
Manufacturing	LPG production	Aldehydes	Secondary	0.39%
Materials processing	Stryene-butadiene copolymer production	Hydrocarbons, remaining unspeciated	Secondary	0.39%
Materials processing	Invar	Nonmethane hydrocarbons, unspeciated	Secondary	0.32%
Manufacturing	Fuel oil #6 production	Hydrocarbons, unspeciated	Secondary	0.31%
Manufacturing	LPG production	Formaldehyde	Secondary	0.29%
Materials processing	Lead	Nonmethane hydrocarbons, unspeciated	Secondary	0.21%
Manufacturing	Fuel oil #6 production	Nonmethane hydrocarbons, unspeciated	Secondary	0.21%
Manufacturing	Natural gas production	Methane	Secondary	0.21%
Manufacturing	LPG production	Ethane	Secondary	0.18%
Manufacturing	CRT tube mfg.	Xylene (mixed isomers)	Primary	0.17%
Manufacturing	LPG production	Pentane	Secondary	0.16%

 Table 3-29.
 Top 99% of the CRT photochemical smog impact score

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Manufacturing	Natural gas production	Benzene	Secondary	0.14%

Table 3-29. T	Fop 99% of the C	RT photochemical	smog impact score
---------------	-------------------------	------------------	-------------------

*Column may not add to 99% due to rounding.

3.3.7.2 Major contributors to the LCD photochemical smog results

Table 3-30 lists the materials that contribute to the top 99% of the LCD life-cycle photochemical smog results. LCD results are dominated by unspeciated hydrocarbon emissions, methane emissions, and benzene emissions from natural gas production in the materials processing life-cycle stage, which together account for about 75% of the total. This natural gas is used as an ancillary material by one LCD monitor/module manufacturer. Other LCD monitor/module manufacturers reported using LNG as a fuel, but not as an ancillary number. A number of other materials and processes contribute more than 1% of the total LCD photochemical smog score, most notably unspeciated, nonmethane hydrocarbon emissions associated with steel production. As noted above under the CRT results, the latter hydrocarbon emissions could occur from any one of various processes (e.g., ore mining, steel production, electricity generation, etc.) wrapped into the steel production inventory.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Materials processing	Natural gas production	Nonmethane hydrocarbons, unspeciated	Secondary	45%
Materials processing	Natural gas production	Methane	Secondary	17%
Materials processing	Natural gas production	Benzene	Secondary	12%
Materials processing	Steel production, cold- rolled, semi-finished	Nonmethane hydrocarbons, unspeciated	Secondary	11%
Manufacturing	LCD monitor/module mfg.	Isopropyl alcohol	Primary	2.5%
Manufacturing	LPG production	Hydrocarbons, unspeciated	Secondary	2.1%
Manufacturing	LPG production	Nonmethane hydrocarbons, unspeciated	Secondary	1.4%
Materials processing	Natural gas production	Hydrocarbons, unspeciated	Secondary	1.2%
Materials processing	Steel production, cold- rolled, semi-finished	Hydrocarbons, unspeciated	Secondary	1.2%
Use	U.S. electric grid	Methane	Model/secondary	1.2%
Manufacturing	Natural gas production	Nonmethane hydrocarbons, unspeciated	Secondary	1.0%
Materials processing	PMMA sheet production	Nonmethane hydrocarbons, unspeciated	Secondary	0.87%
Materials processing	Polycarbonate production	Nonmethane hydrocarbons, unspeciated	Secondary	0.73%
Manufacturing	Natural gas production	Methane	Secondary	0.39%

Table 3-30	Top 99% of the LCD	photochemical	smog impact score
-------------------	--------------------	---------------	-------------------

Tuble e est Tep >> / e el el e protochemical sing impact score				
Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Materials processing	Aluminum production	Nonmethane hydrocarbons, unspeciated	Secondary	0.39%
Materials processing	PET resin production	Nonmethane hydrocarbons, unspeciated	Secondary	0.36%

 Table 3-30. Top 99% of the LCD photochemical smog impact score

*Column may not add to 99% due to rounding.

3.3.7.3 Limitations and uncertainties

Photochemical smog indicators are calculated using the mass of a chemical released to air per functional unit and the chemical-specific partial equivalency factor. The equivalency factor is a measure of the chemical's photochemical oxidant creation potential (POCP) compared to the reference chemical ethylene. As noted in Section 3.1.2.6, photochemical smog impacts are based on partial equivalency because some chemicals cannot be converted into POCP equivalency factors (e.g., nitrogen oxide). The inability to develop equivalency factors for some chemicals is a limitation of the photochemical smog impact assessment methodology.

The CRT impact score for photochemical smog formation is being driven by the process for producing the large amount of LPG used in CRT glass manufacture. However, as discussed in Section 2.3.3.3 and previous subsections of this Section 3.3, the three sets of glass manufacturing energy data received by the CDP were highly variable, making the average glass energy inputs used in the baseline analysis uncertain. Therefore, the emissions of smog forming chemicals from LPG production, which are based on the glass LPG inputs, are also uncertain. CRT glass energy inputs were subjected to a sensitivity analysis. Sensitivity results are discussed in Section 3.4.

The LCD impact score for photochemical smog formation is being driven by the natural gas production process to produce the large amount of LNG used as an ancillary material during LCD monitor/module manufacturing. However, as discussed earlier, only one LCD monitor/module manufacturer reported this use of LNG, which indicates the average LNG inputs used in the LCD inventory may be unduly high. Therefore, the mass of smog forming chemical emissions from the natural gas production process, which are based on the amount of ancillary LNG inputs, may also by unduly high.

The majority of the CRT and LCD photochemical smog results are based on life-cycle inventories from secondary sources and are therefore subject to the limitations and uncertainties associated with secondary data, discussed previously. In particular, see Section 3.3.1.1 for a detailed discussion of limitations and uncertainties in the LPG and natural gas production data.

3.3.8 Acidification

Figure 3.11 presents the CRT and LCD LCIA results for the acidification impact category, based on the impact assessment methodology presented in Section 3.1.2.7. Tables M-19 and M-20 in Appendix M list complete results for the CRT and LCD, respectively. The life-cycle acidification impact indicator result is 5.25 kg of SO₂ equivalents per monitor for the CRT and 2.96 kg of SO₂ equivalents per monitor for the LCD. As might be expected, acidification impacts are greatest in the use stage for both monitor types, due to the emissions of SO_x and NO_x from U.S. power plants. Use stage impacts are more dominant for the CRT (65% of life-cycle acidification impacts) than the LCD (43%) due to the relatively large amount of power consumed during use by the less energy-efficient CRT.



Emissions from manufacturing processes account for 28% of CRT acidification impacts and 38% of LCD acidification impacts. Material processing is responsible for about 7 and 19% of CRT and LCD impacts in this category, respectively. Both technologies receive a slight credit on acidification impacts during the EOL stage due to an energy credit from incineration processes with energy recovery, which offset some of the impacts from electricity production.

3.3.8.1 Major contributors to the CRT acidification results

Table 3-31 lists the materials that contribute to the top 99% of the CRT acidification impacts score and their LCI data type. SO_2 and NO_x emissions from the U.S. electric grid are the two largest contributors to the CRT acidification indicator, together accounting for 63% of the total. SO_x and NO_x emissions from the LPG production process are also significant, contributing about 23%. As noted previously, most of the LPG from this production process is used to manufacture CRT glass, but the mass of LPG inputs is uncertain and the basis of a sensitivity analysis later in this chapter (see Section 3.4).

Other materials that contribute more than one percent of the CRT acidification indicator are SO_2 emissions from invar production, hydrochloric acid emissions from electricity generation during the use stage, and sulfur dioxide emissions from electricity used in manufacturing. Note that most of the LCI data for materials in Table 3-31 are from secondary sources.

Life-cycle stage	Process group	Material	LCI data	Contribution
			type	to impact score*
Use	U.S. electric grid	Sulfur dioxide	Model/secondary	47%
Use	U.S. electric grid	Nitrogen oxides	Model/secondary	16%
Manufacturing	LPG production	Sulfur oxides	Secondary	15%
Manufacturing	LPG production	Nitrogen oxides	Secondary	7.6%
Materials processing	Invar	Sulfur dioxide	Secondary	4.8%
Use	U.S. electric grid	Hydrochloric acid	Model/secondary	1.8%
Manufacturing	Japanese electric grid	Sulfur dioxide	Secondary	1.6%
Manufacturing	U.S. electric grid	Sulfur dioxide	Model/secondary	0.76%
Materials processing	Steel production, cold- rolled, semi-finished	Sulfur dioxide	Secondary	0.73%
Manufacturing	CRT glass/frit manufacturing	Nitrogen oxides	Primary	0.59%
Manufacturing	Japanese electric grid	Nitrogen oxides	Model/secondary	0.54%
Use	U.S. electric grid	Hydrofluoric acid	Model/secondary	0.41%
Materials processing	Aluminum production	Sulfur dioxide	Secondary	0.40%
Materials processing	Polycarbonate production	Nitrogen dioxide	Secondary	0.26%
Manufacturing	U.S. electric grid	Nitrogen oxides	Model/secondary	0.25%
Materials processing	Polycarbonate production	Sulfur dioxide	Secondary	0.23%
Manufacturing	LPG production	Nitrous oxide	Secondary	0.22%

 Table 3-31.
 Top 99% of the CRT acidification impact score

*Column may not add to 99% due to rounding.

3.3.8.2 Major contributors to the LCD acidification impact results

Table 3-32 lists the materials responsible for the top 99% of the LCD acidification impact results and the LCI data type. Sulfur dioxide emissions from the U.S. electric grid during the use stage are the greatest contributor at 31%, followed by NO_x from natural gas production in the materials processing stage. The latter process produces natural gas used by one LCD monitor/module manufacturer as an ancillary material, indicating the LCD monitor/module manufacturing process group is ultimately responsible for this contribution to the impact score. However, as noted previously, only one LCD monitor/module manufacturer reported the ancillary use of LNG. Other LCD monitor/module manufacturers reported using LNG as a fuel, but not as an ancillary material. NO_x , ammonia, hydrofluoric acid, and hydrochloric acid emissions from LCD monitor/module manufacturing data were collected directly by the CDP from manufacturers in Asia.

 NO_x emissions from the U.S. electric grid during the use stage, and SO_x and NO_x emissions from the Japanese electric grid during manufacturing are also among the top contributors to the LCD acidification results. LCI data for these process groups were developed by the CDP from secondary sources.
Life-cycle stage	Process group	Material	LCI data	Contribution
	6 1		type	to impact
				score*
Use	U.S. electric grid	Sulfur dioxide	Model/secondary	31%
Materials processing	Natural gas prod.	Nitrogen oxides	Secondary	15%
Manufacturing	LCD monitor/module mfg.	Nitrogen oxides	Primary	13%
Use	U.S. electric grid	Nitrogen oxides	Model/secondary	10%
Manufacturing	Japanese electric grid	Sulfur dioxide	Model/secondary	9.8%
Manufacturing	LCD monitor/module mfg.	Ammonia	Primary	4.0%
Manufacturing	Japanese electric grid	Nitrogen oxides	Model/secondary	3.2%
Manufacturing	LCD monitor/module mfg.	Hydrofluoric acid	Primary	2.8%
Manufacturing	LCD monitor/module mfg.	Hydrochloric acid	Primary	1.8%
Manufacturing	LPG production	Sulfur oxides	Secondary	1.3%
Use	U.S. electric grid	Hydrochloric acid	Model/secondary	1.2%
Manufacturing	LCD backlight	Nitrogen oxides	Primary	0.70%
Materials processing	Natural gas production	Ammonia	Secondary	0.69%
Materials processing	Natural gas production	Sulfur oxides	Secondary	0.65%
Manufacturing	LPG production	Nitrogen oxides	Secondary	0.65%
Materials processing	Steel production, cold-rolled, semi-finished	Sulfur oxides	Secondary	0.64%
Materials processing	PMMA sheet production	Sulfur oxides	Secondary	0.44%
Manufacturing	Natural gas production	Nitrogen oxides	Secondary	0.35%
Use	U.S. electric gird	Hydrofluoric acid	Model/secondary	0.27%

 Table 3-32. Top 99% to the LCD acidification impact score

*Column may not add to 99% due to rounding.

3.3.8.3 Limitations and uncertainties

Acidification impact characterization is a function of the mass of an acid-forming chemical emitted to air and the acidification potential (AP) equivalency factor for that chemical. The AP equivalency factor is the number of hydrogen ions that can theoretically be formed per mass unit of the pollutant being released compared to SO_2 . This is a full equivalency approach to impact characterization where all substances are addressed in a unified, technical model, which lends more certainty to the characterization results than partial equivalency factors discussed with regard to photochemical smog (Section 3.3.7).

For the CRT, and less so for the LCD, impact results are being driven primarily by SO_2 and NO_x emissions from U.S. power plants during use of the monitor by the consumer. As discussed in Section 3.3.5 and noted above, the U.S. and Japanese electric grid inventories were developed by the CDP from secondary sources. U.S. electric grid emissions of the criteria pollutants, including SO_2 and NO_x , are based on data in the EPA publication, *National Air Quality and Emissions Trends Report, 1997* (EPA, 1998), which were the best data available when the electric grid inventory data was developed and are expected to be reasonably accurate. However, the Japanese electric grid inventory was derived from the U.S. inventory based on the mix of fuels used in Japan. Because Japanese power plants may employ different pollution

control devices, use fuels of different quality, or other factors, their emissions could actually be higher or lower than those reported in the inventory.

LCI data for many of the other primary contributors to the CRT acidification impact category are from existing LCI databases. The limitations and uncertainties associated with these data have been discussed extensively in other subsections of this chapter and pertain here. On the other hand, LCI data for many of the other primary contributors to the LCD acidification indicator results were collected directly by the CDP from manufacturers in Asia. These data are considered to be of better quality since they were collected to meet the goals, objectives and temporal and spatial boundaries of the CDP.

3.3.9 Air Particulates

Figure 3-12 presents the CRT and LCD LCIA results for the air particulates impact category by life-cycle stage, based on the impact assessment methodology presented in Section 3.1.2.8. Tables M-21 and M-22 in Appendix M list complete air particulates results for the CRT and LCD, respectively.



The life-cycle air particulates indicator is 0.30 kg of air particulates per monitor for the CRT and 0.115 kg of air particulates per monitor for the LCD. Recall from Section 3.1.2.8 that air particulates impact results are ideally based on release amounts of particulate matter with average aerodynamic diameter less than 10 micrometers (PM_{10}) to the air. This is the size of particulate matter that is most damaging to the respiratory system. However, as will be shown later in this section, a significant portion of the particulate emissions data for both monitor types do not specify a particulate size. This makes it more difficult to draw conclusions about the relative life-cycle air particulate impacts of the CRT and LCD.

The manufacturing and upstream materials processing stages have almost equal contribution to CRT air particulate impacts, at 45% of the total for the manufacturing stage and 43% of the total for the upstream stages. LCD impacts, on the other hand, are dominated by particulate emissions during the upstream, materials processing stages, which contribute 80% of the total score. Both technologies receive a substantial reduction in life-cycle air particulate

impacts at EOL due to an energy credit from incineration with energy recovery. The energy credit, which is from incineration with energy recovery, is applied to electric power production where it offsets some particulate emissions that would otherwise occur from electrical power production.

3.3.9.1 Major contributors to the CRT air particulates impact results

Table 3-33 lists the materials that contribute to the top 99% of the CRT air particulates impact score and their LCI data type. PM emissions from LPG production are the single largest contributor to the overall score, at 43% of the total. This LPG is primarily an energy source in CRT glass manufacturing, indicating the glass/frit process group is the ultimate source of these air particulate emissions. As noted previously, CRT glass energy inputs are the subject of a sensitivity analysis, discussed in Section 3.4.

Other major contributors to the CRT air particulates impact results are PM emissions from the steel production process group, PM_{10} emissions from the U.S. electric grid during the use stage, and PM emissions from aluminum production processes. Note that the inventories for steel and aluminum production combine data from the raw materials extraction, materials manufacture, and electricity generation processes. The PM emissions reported for the material production process group could be from any one of these individual processes, but particulate matter emissions are most often associated with combustion processes.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Manufacturing	LPG production	PM	Secondary	43%
Materials processing	Steel production, cold-rolled, semi-finished	PM	Secondary	35%
Use	U.S. electric grid	PM-10	Model/secondary	19%
Materials processing	Aluminum production	PM	Secondary	3.0%

 Table 3-33.
 Top 99% of the CRT air particulates impact score

*Column adds to greater than 99% due to an offset of emissions from incineration with energy recovery at EOL.

As shown in Table 3-33, the impact scores associated with the LPG, steel, and aluminum production process groups—82% of CRT air particulates impacts—are based on emissions of PM instead of emissions of PM₁₀. This could be a matter of different terminology used in the secondary data sets for these process groups (that is, PM is used to represent PM₁₀), or it could represent a broader class of particulate emissions, of which PM₁₀ emissions would be a subset. If the latter case is true, it is likely that CRT air particulate impacts are overstated.

3.3.9.2 Major contributors to the LCD air particulates impact results

Table 3-34 lists the materials that contribute to the top 99% of the LCD air particulates impact score and their LCI data type. PM emissions from steel production are the largest contributor to the overall score, followed by PM emissions from natural gas production. Natural gas from this process supplies the LNG used as an ancillary material by one LCD monitor/module manufacturer.

Other major contributors to the LCD air particulates impact results are PM_{10} emissions from the U.S. electric grid during the use stage and from the Japanese electric grid during manufacturing, and PM emissions from LPG production. LPG from the latter process supplies energy to the LCD glass manufacturing process.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Materials processing	Steel production, cold-rolled, semi-finished	РМ	Secondary	45%
Materials processing	Natural gas production	PM	Secondary	25%
Use	U.S. electric grid	PM-10	Model/secondary	19%
Manufacturing	Japanese electric grid	PM-10	Model/secondary	5.9%
Manufacturing	LPG production	PM	Secondary	5.4%

Table 3-34. Top 99% of the LCD air particulates impact score

*Column adds to greater than 99% due to an offset of emissions from incineration with energy recovery at EOL.

As shown in Table 3-34, the impact scores associated with the steel, natural gas, and LPG production process groups—roughly 75% of LCD air particulates impacts—are based on emissions of PM instead of emissions of PM_{10} . As with the CRT, air particulate impacts should be based on PM_{10} emissions, indicating LCD air particulate impacts may be overstated.

3.3.9.3 Limitations and uncertainties

The CDP LCIA methodology for air particulates is based on emissions of PM_{10} to air, which is the size of particulate matter that is most damaging to the respiratory system. However, as noted in Tables 3-33 and 3-34, the majority of the CRT and LCD impacts were calculated from emissions of "PM" rather than PM_{10} . This could be a matter of different terminology used in the secondary data sets for these process groups, or it could represent a broader class of particulate emissions, of which PM_{10} emissions would be a subset. If the latter case is true, it is likely that both the CRT and LCD air particulate impacts are overstated.

The LCI data for all of the major contributors to both the CRT and LCD were either developed by the CDP from secondary sources (e.g., the U.S. and Japanese electric grids) or are from secondary LCI data sets (e.g., the fuel and upstream materials production processes). The limitations and uncertainties associated with these data have been discussed in other subsections of this chapter and pertain here. Note that U.S. electric grid emissions of the criteria pollutant PM₁₀ are based on data in the EPA publication, *National Air Quality and Emissions Trends Report, 1997* (EPA, December, 1998, EPA/454/R-98-016), and are expected to be reasonably accurate.

Finally, the amount of LPG used to produce CRT glass, which is ultimately driving the CRT air particulates results, is also uncertain due to the large variability in CRT glass energy inputs received from glass manufacturers. See Section 3.4 for a sensitivity analysis of CRT glass energy inputs.

3.3.10 Water Eutrophication

Figure 3-13 presents the CRT and LCD LCIA results for the water eutrophication impact category by life-cycle stage, based on the impact assessment methodology presented in Section 3.1.2.9. Tables M-23 and M-24 in Appendix M are complete results for the CRT and LCD, respectively.



The life-cycle water eutrophication indicators are 0.048 kg of phosphate equivalents for the CRT and 0.050 kg of phosphate equivalents for the LCD. Results for both the CRT and LCD are completely dominated by emissions from the manufacturing stage, which accounts for 99% of the indicator for both technologies. Both technologies have negative scores at the end-of-life due to incineration with energy recovery. The energy recovery offsets some of the water emissions from the electricity generation inventory included in the incineration data set.

3.3.10.1 Major contributors to the CRT water eutrophication impact results

Table 3-35 lists the materials that contribute to the top 99% of the CRT water eutrophication results. Together, chemical oxygen demand (COD) and ammonia ions from the LPG production process group account for about 91% of the total score. Most of the LPG from this process is used as an energy source in CRT glass manufacturing (see Section 3.4 for the sensitivity analysis of CRT glass energy inputs). Emissions of nitrogen, COD and phosphorus from the CRT tube manufacturing process group contribute about seven percent of the CRT water eutrophication impacts. COD and other nitrogen emissions from steel production are the remaining top contributors to the CRT eutrophication score. LPG and steel production data are from secondary sources, while the CRT tube manufacturing outputs are primary data collected by the CDP.

	Table 5-55. Top 7770 of the CKT water eutrophication impact score					
Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*		
Manufacturing	LPG production	COD	Secondary	72%		
Manufacturing	LPG production	Ammonia ions	Secondary	19%		
Manufacturing	CRT tube manufacturing	Nitrogen	Primary	6.3%		
Materials processing	Steel production, cold-rolled, semi-finished	Other nitrogen	Secondary	0.37%		
Materials processing	Steel production, cold-rolled, semi-finished	COD	Secondary	0.33%		
Manufacturing	CRT tube manufacturing	COD	Primary	0.33%		
Manufacturing	CRT tube manufacturing	Phosphorus (yellow or white)	Primary	0.32%		

 Table 3-35.
 Top 99% of the CRT water eutrophication impact score

*Column may not add to 99% due to rounding.

3.3.10.2 Major contributors to the LCD water eutrophication impact results

Table 3-36 lists the materials that contribute to the top 99% of the LCD water eutrophication results. Like the CRT, the LCD water eutrophication indicator is driven by emissions from a single process group, in this case, LCD monitor/module manufacturing. Together, emissions of nitrogen and phosphorus from that process account for 94% of the total score. The LPG production process is the next largest contributor to the LCD water eutrophication score, where water releases of COD and ammonia ions account for more than 4% of the total.

 Table 3-36. Top 99% to the LCD water eutrophication impact score

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Manufacturing	LCD monitor/module mfg.	Nitrogen	Primary	67%
Manufacturing	LCD monitor/module mfg.	Phosphorus (yellow or white)	Primary	27%
Manufacturing	LPG production	COD	Secondary	3.4%
Manufacturing	LPG production	Ammonia ions	Secondary	0.88%
Manufacturing	LCD panel components	Phosphorus (yellow or white)	Primary	0.48%
Materials processing	PMMA sheet production	Ammonia	Secondary	0.40%

*Column may not add to 99% due to rounding.

3.3.10.3 Limitations and uncertainties

Eutrophication (nutrient enrichment) impacts are calculated from the mass of a chemical released directly to surface water and the chemical's eutrophication potential (EP). The EP is a partial equivalency factor derived from the ratio of nitrogen and phosphorus in the average composition of algae compared to the reference compound phosphate (see Section 3.1.2.9). As a partial equivalency approach, only a subset of substances can be converted into equivalency

factors, which is a limitation of this LCIA methodology. However, the methodology does take into account nitrogen and phosphorus, which are the two major limiting nutrients of importance to eutrophication.

CRT water eutrophication results are dominated by LCI data from secondary sources, and are therefore subject to the limitations and uncertainties associated with secondary data. Furthermore, these results are ultimately due to the large amount of LPG reported to be used as a fuel in LPG glass production. Because of the large degree of variability in glass energy data received from three CRT glass manufacturers, CRT glass energy inputs are also uncertain and the subject of a sensitivity analysis (see Section 3.4). LCD results, on the other hand, are driven almost entirely by primary LCI data from the manufacturing life-cycle stage, which were collected to meet the goals, objectives, and temporal and geographic boundaries of the CDP and are therefore considered to be of better quality.

3.3.11 Water Quality

3.3.11.1 Biological oxygen demand (BOD)

Figure 3-14 presents the CRT and LCD LCIA results for the BOD water quality impacts category by life-cycle stage, based on the impact assessment methodology presented in Section 3.1.2.10. Complete results are listed in Tables M-25 (CRT) and M-26 (LCD) in Appendix M.



During the life-cycle of a 17" CRT monitor, 0.195 kg of BOD are released to surface water. The life-cycle of a functionally equivalent 15" LCD results in 0.0283 kg of BOD surface water releases. As shown in Figure 3-14, BOD impacts for both monitor types are driven by surface water releases in the manufacturing stage, which contribute 100% of CRT impacts and 99% of LCD impacts. Note that small BOD impacts also occur in the upstream, materials processing life-cycle stage for both monitor types. These are almost entirely offset by negative BOD values at end of life due to the offset of electric grid emissions when the monitors are incinerated with energy recovery. Note also that there are no BOD emissions from the U.S. electric grid during the use stage. The incineration inventory is a secondary data set that

contains a different electric grid inventory than the U.S. electric grid inventory developed by the CDP.

Table 3-37 lists the materials responsible for the top 99% of the CRT BOD impacts. CRT impacts in this category are driven by BOD releases from the LPG production process, most of which is used to make LPG employed as fuel in CRT glass manufacturing. BOD releases from CRT tube manufacturing also contribute a small percentage to the total score.

Table 5-57. Top 7770 of the CKT DOD impact score					
Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score	
Manufacturing	LPG production	BOD	Secondary	96%	
Manufacturing	CRT tube manufacturing	BOD	Primary	3.3%	

Table 3-37.	Тор 99%	of the CRT	BOD im	pact score
-------------	---------	------------	--------	------------

Table 3-38 lists the materials that contribute to the top 99% of the LCD BOD impacts. As shown in the table, LCD impacts are slightly more distributed among processes than CRT impacts, with BOD releases from four processes or process groups making up the list of top contributors. Note that BOD releases from LPG production (most of which is used to make LPG for LCD glass manufacturing) are much less for the LCD than the CRT, even though the LCD glass manufacturing inventory was derived from the CRT glass manufacturing inventory. This is because the CRT contains approximately ten times more glass than the LCD.

	Table 5-56. Top 7776 of the LCD BOD impact score					
Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score		
Manufacturing	LCD monitor/module mfg.	BOD	Primary	61%		
Manufacturing	LPG production	BOD	Secondary	32%		
Manufacturing	LCD panel components	BOD	Primary	4.7%		
Materials processing	Natural gas production	BOD	Secondary	0.99%		

Table 3-38. Top 99% of the LCD BOD impact score

3.3.11.2 Total suspended solids (TSS)

Figure 3-15 presents the CRT and LCD LCIA results for the TSS water quality impacts category by life-cycle stage, based on the impact assessment methodology presented in Section 3.1.2.10. Tables M-27 and M-28 in Appendix M list complete results for the CRT and LCD, respectively.

The life-cycle TSS impact indicator is 0.874 kg of TSS for the CRT and 0.0615 kg of TSS for the LCD. TSS impacts for both monitor types are driven by the manufacturing stage, where 99 and 94% of impacts occur for the CRT and LCD, respectively. TSS impacts also occur in the upstream, materials processing life-cycle stage for both monitor types. Both technologies receive a credit on TSS impacts at EOL due to an offset of electric grid emissions when the monitors are incinerated with energy recovery.



Table 3-39 presents the major contributors to the CRT TSS indicator and lists the LCI data type. As with many other impact categories, the LPG production process is the single largest contributor to the CRT TSS indicator, accounting for 97% of the total score. Most of the LPG from this process is used as a fuel to produce CRT glass, but CRT energy inputs are uncertain and evaluated in a sensitivity analysis in Section 3.4. TSS surface water releases from the CRT glass/frit process group, CRT tube manufacturing, and fuel oil #6 production are also top contributors to the CRT TSS score. However, the contribution of these processes or process groups is small compared to that of the LPG production process.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score
Manufacturing	LPG production	Suspended solids	Secondary	97%
Manufacturing	CRT glass/frit mfg.	Suspended solids	Primary	0.83%
Manufacturing	CRT tube manufacturing	g Suspended solids	Primary	0.53%
Manufacturing	Fuel oil # 6 production	Suspended solids	Secondary	0.33%

Table 3-39. Top 99% of the CRT TSS impact score

Table 3-40 presents the top contributors to the LCD TSS impact score. Like the CRT results discussed above, TSS surface water releases from the LPG production process are responsible for the majority of LCD TSS impacts. LPG from this production process is used to produce LCD glass. Note that the actual mass of TSS releases from the LCD-related process is much smaller than those from the CRT-related process. This is because the LCD only uses about 10% as much glass as the CRT. TSS releases from the LCD monitor/module process group also account for a sizeable percentage of LCD TSS impacts.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*		
Manufacturing	LPG production	Suspended solids	Secondary	66%		
Manufacturing	LCD monitor/module mfg.	Suspended solids	Primary	25%		
Materials processing	PMMA sheet production	Suspended solids	Secondary	2.2%		
Materials processing	Natural gas production	Suspended solids	Secondary	2.0%		
Materials processing	Steel production, cold-rolled, semi-finished	Suspended solids	Secondary	1.6%		
Materials processing	Aluminum production (all virgin)	Suspended solids	Secondary	1.1%		
Manufacturing	LCD panel components	Suspended solids	Primary	1.0%		

 Table 3-40.
 Top 99% of the LCD TSS impact score

*Column may not add to 99% due to rounding.

3.3.11.3 Limitations and uncertainties

Both BOD and TSS indicators are calculated using a loading approach (i.e., the impact score is based on the inventory amounts) and are therefore highly sensitive to inventory data quality. CRT impact results are driven almost entirely by the LPG production inventory (a secondary data set) and are therefore subject to the limitations and uncertainties associated with secondary data. In particular, see Section 3.3.2.3 for a detailed discussion of LPG production data quality. In addition, note that LPG production impacts are almost all due to the large amount of LPG reported to be used as a fuel in CRT glass manufacturing. As note previously, CRT glass energy inputs are uncertain and are evaluated in a sensitivity analysis (see Section 3.4).

LCD impact results, on the other hand, are driven by LCI data from both primary and secondary sources and are therefore considered to be of somewhat better quality than the CRT results. However, a significant percentage of LCD water quality impacts also come from the large amount of LPG used as a fuel input to LCD glass manufacturing. These energy inputs are also uncertain and are evaluated in the sensitivity analysis in Section 3.4.

3.3.12 Radioactivity

Figure 3-16 presents the CRT and LCD LCIA results for the radioactivity impact category by life-cycle stage, based on the impact assessment methodology presented in Section 3.1.2.11. Complete CRT and LCD results are presented in Tables M-29 and M-30 in Appendix M, respectively.

The life-cycle radioactivity indicator is 38.5 million Bequerels (Bq) for the CRT and 12.2 million Bq for the LCD. Radioactivity impacts are driven by radioactive emissions from the upstream, materials processing stage for both monitor types, which contributes 99% of CRT life-cycle impacts and 98% of LCD life-cycle impacts. This result was unforeseen, since one might expect the majority of radioactive emissions to occur from the use stage, due to electricity generation at nuclear power plants. As it turns out, radioactivity impacts are being driven by data for nuclear fuel reprocessing that are included in the electric grid inventories for the steel, invar (an alloy of nickel and ferrite), and ferrite production process groups.



The LCIs for steel, invar, and ferrite were obtained from two databases developed by the former *Ecobilan Group*, an LCA consulting firm that was previously headquartered in France (see Section 2.2.1.1 for a discussion of how these databases were selected). Per *Ecobilan*, the ferrite inventory contains older data that may include radioactive emissions from electricity use. For the steel and nickel inventories, the source of both is site data in Europe, with the radioactive emissions coming from electricity in Europe, where nuclear fuel is reprocessed. In fact, the electricity data are from France for both materials, and France is one of the few countries (including Japan and the United Kingdom) that reprocesses nuclear fuel (Glazebrook, 2001). Therefore, the radioactivity impacts calculated from these inventories are more representative of impacts from countries that reprocess nuclear fuel than impacts from countries that do not.

To further illustrate this point, Tables 3-41 and 3-42 lists the materials and process groups that contribute to the top 99% of the CRT and LCD radioactivity indicator results, respectively. As shown in the tables, radioactivity impacts for both monitor types are driven by releases of plutonium-241 from steel (both monitor types), invar (CRT), and ferrite (CRT) production. Plutonium-241 is a byproduct of fuel reprocessing. Xenon -133 releases from the U.S. electric grid contribute slightly to the LCD radioactivity impacts and to a lesser degree to the CRT total impacts. Note that the actual amount of radioactivity from Xenon-133 is greater for the CRT than the LCD, but contributes a smaller percent of total impacts.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Materials processing	Steel production, cold- rolled, semi-finished	Plutonium-241 (isotope)	Secondary	62%
Materials processing	Invar	Plutonium-241 (isotope)	Secondary	18%
Materials processing	Ferrite	Plutonium-241 (isotope)	Secondary	17%
Use	U.S. electric grid	Xenon-133 (isotope)	Model/secondary	0.81%
Materials processing	Steel production, cold- rolled, semi-finished	Plutonium-240 (isotope)	Secondary	0.27%
Materials processing	Steel production, cold- rolled, semi-finished	Cesium-135 (isotope)	Secondary	0.24%

 Table 3-41. Top 99% of the CRT radioactivity impact score

*Column may not add to 99% due to rounding.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*			
Materials processing	Steel production, cold- rolled, semi-finished	Plutonium-241 (isotope)	Secondary	96%			
Use	U.S. electric grid	Xenon-133 (isotope)	Model/secondary	0.95%			
Manufacturing	Japanese electric grid	Xenon-133M (isotope)	Secondary	0.54%			
Materials processing	Steel production, cold- rolled, semi-finished	Plutonium-240 (isotope)	Secondary	0.42%			
Materials processing	Steel production, cold- rolled, semi-finished	Cesium-135 (isotope)	Secondary	0.38%			

 Table 3-42. Top 99% of the LCD radioactivity impact score

*Column may not add to 99% due to rounding.

Most of the radioactivity impacts are based on LCI data from secondary sources and are therefore subject to the limitations and uncertainties in secondary data, discussed previously. In addition, because radioactivity impacts are being driven by radioactive emissions from fuel reprocessing in France, they may not be representative of radioactivity impacts elsewhere. However, most of the CRT and LCD primary manufacturing data were collected from companies in Japan, where fuel reprocessing also occurs. For example, if Japanese CRT and LCD monitor and/or components manufacturers purchase steel from Japanese steel mills, the radioactivity emissions from electricity used to manufacture the steel could be similar. Japan ranked second in worldwide steel production in 2000 behind Mainland China, and third in 1999 behind Mainland China and the United States (IISI, 2001).

Note that the Japanese electric grid, which is linked to CRT and LCD production inventories, was developed from the U.S. electric grid inventory and therefore does not account for radioactive emissions from fuel reprocessing. This means that radioactive impacts from Japanese manufacturing processes that consume electricity are understated. For example, electricity used in the CRT glass/frit process group was the ninth largest contributor to the CRT energy use score, but the inventory for this process group does not account for fuel reprocessing emissions.

3.3.13 Potential Human Health Impacts

As discussed in Section 3.1.2.12, human health impacts included in the scope of this LCA are chronic (repeated dose) effects, including non-carcinogenic and carcinogenic effects to both workers and the public, and aesthetics. (Although not a health effect *per se*, aesthetics pertains to human welfare.)

Chronic health effect (cancer and noncancer) impacts are calculated using the scoring of inherent properties approach where an impact score is based on the inventory amount weighed by a hazard value (HV). The HV represents the chronic toxicity of a specific material (see Table K-8 in Appendix K for a list of toxicity values used to calculate hazard values). In this manner the inventory amount (the toxic chemical input amount for occupational health effects, and the output amount for public health effects) is used as a surrogate for exposure, while the hazard value represents the inherent toxicity of the chemical for chronic exposure.

The CDP human health effects LCIA methodology does not consider the fate and transport of a toxic chemical in the environment, nor does it evaluate the potential for actual exposures to occur. LCI data do not have the temporal and spatial specificity needed to estimate potential dose rates, for example, nor do they contain information on engineering controls used in an occupational setting to reduce exposure. [It should be noted that more sophisticated models for evaluating human health effects in an LCA framework are being developed that use a multimedia fate, multi-pathway human exposure, and toxicological potency approach (Bare, 1999). However, such models are less comprehensive in terms of the number of chemicals for which there are data.] The limitations and uncertainties in the health effects scores are discussed further below, following the presentation of results.

3.3.13.1 Chronic occupational health effects

Figure 3-17 presents the CRT and LCD LCIA results by life-cycle stage for the chronic occupational health effects impact category, based on the impact assessment methodology presented in Section 3.1.2.12. Complete CRT and LCD results are presented in Tables M-31 and M-32 in Appendix M.



The life-cycle chronic occupational health effects indicator is 934 tox-kg per functional unit for the CRT, and 696 tox-kg per functional unit for the LCD. As shown in the figure, the total score is dominated by toxic chemical inputs to the manufacturing stage, which account for 98% and 96% of CRT and LCD impacts in this category, respectively. This result was expected since inputs to the other life-cycle stages tend to be raw materials (e.g., ores, coal, etc., for the materials processing and use life-cycle stages) or finished products (e.g., the monitors themselves for the EOL stage) that are not classified as toxic materials (see Table K-9 in Appendix K for a l

list of materials excluded from the toxic classification). Both the CRT and LCD receive negative chronic occupational health effects scores at end of life due to the offset of electric grid emissions when the monitors are incinerated with energy recovery.

Table 3-43 lists the materials responsible for the top 99% of the CRT chronic occupational health effects score and the LCI data type. LCI data for most of the top contributors are primary data collected from manufacturers by the CDP. In general, these data are expected to be of better quality (for the purposes of the CDP) than data from secondary sources, since they were collected to meet the goals and scope of the CDP.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Manufacturing	CRT glass/frit mfg.	Liquified petroleum gas	Primary	75%
Manufacturing	PWB manufacturing	Sulfuric acid	Primary	13%
Manufacturing	CRT tube manufacturing	Sulfuric acid	Primary	4.1%
Use	U.S. electric grid	Natural gas	Model/secondary	3.0%
Manufacturing	CRT glass/frit mfg.	Barium carbonate	Primary	1.8%
Use	U.S. electric grid	Petroleum (in ground)	Model/secondary	0.81%
Manufacturing	CRT tube manufacturing	Fuel oil # 6	Primary	0.79%

 Table 3-43. Top 99% of the CRT chronic occupational health effects score

*Column may not add to 99% due to rounding.

LPG inputs to the glass/frit process group, primarily from CRT glass manufacturing, contribute 75% of the CRT impacts in this category. The high impact score for LPG is mainly due to the large amount of LPG inputs to the glass/frit process group (351 kg/functional unit), which results in a high score when multiplied by the HV. No toxicity data were available for LPG. Therefore, it was assigned default HVs of one for both cancer and noncancer effects (total HV=2), representative of mean cancer and noncancer toxicity values. As noted previously, glass manufacturing energy data are uncertain and therefore evaluated in a sensitivity analysis (see Section 3.4).

Sulfuric acid used in PWB manufacturing is the next greatest contributor to the CRT chronic occupational health effects results (13%) followed by sulfuric acid used in CRT tube manufacturing (4.1%). The sulfuric acid HV is based on an inhalation NOAEL of 0.1 mg/m³, which is significantly lower (and therefore more toxic) than the geometric mean inhalation NOAEL of 68.7 mg/m³. Consequently, sulfuric acid impacts are driven more by its inherent toxicity for noncancer effects than the input amounts (0.18 kg per functional unit for PWB manufacturing and 0.056 kg per functional unit for CRT tube manufacturing). Sulfuric acid has

no cancer slope factor and an IARC weight of evidence (WOE) classification of 3 (not classifiable for carcinogenicity), and therefore received an HV of zero for cancer effects.

Natural gas and petroleum used as fuels in the U.S. electric grid, barium carbonate used in the CRT glass/frit process group and fuel oil #6 used to manufacture the CRT tube round out the top contributors to the CRT chronic occupational health effects score. Barium carbonate has no cancer slope factor and an EPA cancer WOE of D (not classifiable), and therefore has an HV of zero for cancer effects. However, its oral NOAEL is 0.21 mg/kg-day compared to the geometric mean oral NOAEL of 11.9 mg/kg-day, which results in an HV of 57 for noncancer effects. Therefore, like sulfuric acid, the barium carbonate impacts are driven more by its inherent toxicity than the input amount (0.297 kg per functional unit). No specific toxicity data were available for natural gas, petroleum, or fuel oil #6; consequently they were assigned a default HV of one for both cancer and noncancer effects (total HV=2).

Table 3-44 lists the materials that contribute to the top 99% of the LCD chronic occupational health effects score. Like the CRT, LCI data for most of the top contributors are primary data collected from manufacturers by the CDP, and are therefore expected to be of generally better quality (for the purposes of the CDP) than data from secondary sources.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Manufacturing	LCD monitor/module mfg.	Liquified natural gas	Primary	57%
Manufacturing	LCD monitor/module mfg.	Sulfuric acid	Primary	23%
Manufacturing	PWB manufacturing	Sulfuric acid	d Primary 8.0%	
Manufacturing	LCD glass manufacturing	Liquified petroleum gas	Primary	4.7%
Manufacturing	LCD monitor/module mfg.	Phosphine	Primary	1.8%
Manufacturing	LCD panel components	Sulfuric acid	Primary	1.6%
Use	U.S. electric grid	Natural gas	Model/secondary	1.5%
Manufacturing	LCD monitor/module mfg.	Dimethylsulfoxide	Primary	1.1%
Manufacturing	LCD monitor/module mfg.	Ethanolamine	Primary	0.62%

 Table 3-44. Top 99% of the LCD chronic occupational health effects score

*Column may not add to 99% due to rounding.

As shown in the table, LCD impacts in this category are dominated by LNG used in LCD monitor/module manufacturing. The high impact score for LNG is primarily due to the large amount of ancillary LNG inputs (194 kg/functional unit) in the LCD monitor/module manufacturing inventory, which results in a high score when multiplied by the HV. No toxicity data were available for LNG. Therefore, it was assigned a default, mean HV of one for both cancer and noncancer effects (total HV=2). As noted previously, only one of seven LCD module/monitor manufacturers reported using LNG as an ancillary material. Therefore, the total score for LCD chronic occupational health effects may not be representative of the industry as a whole. If we remove this application of LNG from the LCD inventory, the LCD occupational health effects result is reduced by 58 percent, from 683 tox-kg per monitor to 288 tox-kg per monitor. Note, however that other LCD monitor/module manufacturers did report using LNG as a fuel.

Sulfuric acid used in three process groups (LCD module/monitor manufacturing, PWB manufacturing, and LCD panel components) accounts for another 33% of the LCD chronic occupational health effects score. As discussed above for the CRT, sulfuric acid has a relatively

low toxicity value, and therefore a high HV, which results in a high impact score for a small input amount. The LCD module/monitor manufacturing process group has the highest impact score for sulfuric acid because it has the greatest input amount (0.229 kg per functional unit).

The remaining top contributors to the LCD chronic occupational health effects score are LPG used in LCD glass manufacturing; and phosphine, dimethylsulfoxide, and ethanolamine used in LCD monitor/module manufacturing; and natural gas used as a fuel by the U.S. electric grid. As discussed earlier, LCD glass energy inputs are uncertain and evaluated in a sensitivity analysis in Section 3.4. The LPG score is based on default HVs (representative of the geometric mean toxicity values) for both cancer and noncancer effects, since no toxicity data were available for LPG.

The phosphine score is driven by its low oral NOAEL value (0.026 mg/kg-day), which is significantly lower than the geometric mean value of 68.7 mg-kg-day. Thus, due to its inherently high toxicity, a relatively small input of phosphine (in this case, 0.027 kg per functional unit) results in a relatively high chronic occupational health effects score. No slope factors or cancer WOE classifications were found for phosphine, indicating a default HV of one was used, which is far outweighed by the non-cancer hazard value.

Dimethylsulfoxide is less toxic than phosphine (oral LOAEL =1.0 mg/kg-day), but has a greater input amount (0.066 kg per functional unit). However, phosphine's greater toxicity outweighs the greater input amount for dimethylsulfoxide, resulting in a higher impact score for phosphine.

No specific toxicity data were available for natural gas; consequently it was assigned a default HV of one for both cancer and noncancer effects (total HV=2).

3.3.13.2 Chronic public health effects

Figure 3-18 presents the CRT and LCD LCIA scores by life-cycle stage for the chronic public health effects category, based on the impact assessment methodology presented in Section 3.1.2.12. Complete results are presented in Tables M-33 and M-34 in Appendix M, respectively.

The life-cycle chronic public health effects score is 1,980 tox-kg per functional unit for the CRT and 902 tox-kg per functional unit for the LCD. As shown in the figure, the CRT score is dominated by toxic chemical outputs from electricity generation in the use stage, which



account for almost 84% of CRT impacts in this category. To a lesser degree, LCD chronic public health effect impacts are also driven by emissions from electricity generation in the use stage, which account for more than 68% of the total. Note that the ratio of CRT to LCD use stage public health impacts is the same as the ratio of CRT to LCD use stage electricity consumption (634 kWh/life for the CRT to 237 kWh/life for the LCD).

The materials processing stage contributes almost 12% of CRT chronic public health effect impacts and almost six percent of LCD impacts. The manufacturing life-cycle stage is responsible for five and 26% of CRT and LCD impacts in this category, respectively. Both monitors receive very small public chronic health effects scores at end of life. This is because most public health effect impacts from CRT and LCD recycling and disposal processes are offset by a credit on electric grid emissions when the monitors are incinerated with energy recovery.

Table 3-45 presents the materials that contribute to the top 99% of the CRT chronic public health effects score. As shown in the table, SO₂ emissions from a number of different process groups almost completely dominate CRT impacts in this category, accounting for more than 98% of the total. All of the SO₂ LCI data shown in the table are either from secondary data sets not developed specifically for the CDP, or from the electric grid inventories developed from secondary sources for this project. Sulfur dioxide has a relatively high HV based on an inhalation NOAEL of 0.104 mg/m³, compared to the geometric mean inhalation NOAEL of 68.7 mg/m³. In addition, from a mass loading perspective, SO₂ emissions were the second largest contributor to CRT life-cycle air pollutant emissions, exceeded only by emissions of carbon dioxide (CO₂). Carbon dioxide is not classified as toxic, and therefore does not contribute to the human health effects scores.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Use	U.S. electric grid	Sulfur dioxide	Model/secondary	83%
Materials processing	Invar	Sulfur dioxide	Secondary	8.3%
Manufacturing	Japanese electric grid	Sulfur dioxide	Model/secondary	2.9%
Manufacturing	U.S. electric grid	Sulfur dioxide	Model/secondary	1.3%
Materials processing	Steel production, cold- rolled, semi-finished	Sulfur dioxide	Secondary	1.3%
Materials processing	Aluminum production	Sulfur dioxide	Secondary	0.70%
Materials processing	Polycarbonate production	Sulfur dioxide	Secondary	0.40%
Manufacturing	LPG production	Carbon monoxide	Secondary	0.29%
Materials processing	Lead production	Sulfur dioxide	Secondary	0.23%

Table 3-45. Top 99% of the CRT chronic public health effects score

*Column may not add to 99% due to rounding.

Most of the sulfur dioxide emissions that contribute to the CRT chronic public health effects score are from the combustion of fossil fuels used to generate electricity. For example, the electricity required to power the monitor during the use stage accounts for the vast majority of SO₂ emissions and 83% of the CRT chronic public health effects score. Sulfur dioxide emissions from electricity consumed in the United States and Japan during the manufacturing life-cycle stage account for another 4.2% of the total score. Much of the SO₂ emissions reported

in secondary data sets for the materials processing life-cycle stage may also be from electricity generation since many of these data also contain an electric grid inventory.

Table 3-46 lists the materials that contribute to the top 99% of the LCD chronic public health effects score and the LCI data type. LCD impacts in this category are also dominated by SO₂ emissions, which are responsible for roughly 93% of impacts. Like the CRT, most of these emissions occur from electricity generation, either during the use stage (68%) or manufacturing (21%). As noted previously, SO₂ has a relatively high HV due to its low toxicity value (inhalation NOAEL = 0.104 mg/m³). Similar to the CRT, from a mass loading perspective, SO₂ emissions were the second largest contributor to LCD life-cycle air pollutant emissions, exceeded only by emissions of CO₂.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Use	U.S. electric grid	Sulfur dioxide	Model/secondary	68%
Manufacturing	Japanese electric grid	Sulfur dioxide	Model/secondary	21%
Manufacturing	LCD monitor/module mfg.	Phosphine	Primary	3.2%
Materials processing	Steel production, cold- rolled, semi-finished	Sulfur dioxide	Secondary	1.4%
Materials processing	PMMA sheet production	Sulfur dioxide	Secondary	0.96%
Materials processing	Natural gas production	Methane	Secondary	0.78%
Materials processing	Natural gas production	Benzene	Secondary	0.59%
Materials processing	Aluminum production	Sulfur dioxide	Secondary	0.57%
Materials processing	Natural gas production	Carbon monoxide	Secondary	0.53%
Materials processing	Polycarbonate production	Sulfur dioxide	Secondary	0.49%
Manufacturing	LCD monitor/module mfg.	Phosphorus (yellow or white)	Primary	0.38%
Manufacturing	U.S. electric grid	Sulfur dioxide	Model/secondary	0.35%

 Table 3-46. Top 99% of the LCD chronic public health effects score

*Column may not add to 99% due to rounding.

Other top contributors to the LCD chronic public health effects score include phosphine and phosphorus from LCD monitor/module manufacturing, and methane, benzene, and carbon monoxide from natural gas production. As noted above in the section on chronic occupational health effects, the phosphine score is driven by its low oral NOAEL value (0.026 mg/kg-day), which is significantly lower than the geometric mean value of 68.7 mg-kg-day, resulting in a high HV. Thus, a relatively small output of phosphine (in this case, air emissions of 0.063 kg per functional unit) results in a relatively high chronic public health effects score.

The benzene and phosphorus chronic health effects scores are also driven more by their toxicity than the output amounts. Benzene is a known human carcinogen (EPA WOE Class A) that also causes noncancer health effects. The HV for benzene is based on its oral slope factor [0.055 (mg/kg-day)⁻¹] and its inhalation NOAEL for noncancer effects (1.15 mg/m³), which together result in a high HV. (Benzene also has an *inhalation* slope factor and an *oral* NOAEL value, but these yield lower hazard values when compared to the geometric mean values.)

Phosphorus has an EPA WOE classification of D (not classifiable as to human carcinogenicity), but has a low oral NOAEL value (0.015 mg/kg-day), which also gives it a high HV.

No toxicity data were available for methane. Therefore, it received a default HV of one for both cancer and noncancer effects (HV=2 total). The HV for carbon monoxide is based on an inhalation LOAEL of 55 mg/m³.

3.3.13.3 Chronic public health effect scores modified to exclude sulfur dioxide

Because the chronic public health effects scores for both the CRT and the LCD are dominated by SO_2 emissions, a secondary analysis was run to identify the top contributors to public health impacts when SO_2 emissions are excluded from the inventories. Results of this analysis may be more useful to manufacturers seeking to identify problematic toxic chemicals within their own manufacturing processes.

Figure 3-19 presents the CRT and LCD chronic public health effects scores by life-cycle stage when SO_2 emissions are excluded from the inventories. Under this scenario, the CRT score is reduced almost 99% from 1980 tox-kg to 26 tox-kg per functional unit, and the LCD score is reduced about 93% from 902 tox-kg to 61 tox-kg per functional unit. Note that these scores should not be used to evaluate which monitor type has higher overall impacts in this category, but they are useful for identifying life-cycle improvement opportunities that were previously obscured by SO_2 impacts.



With SO₂ emissions removed from the inventories, chronic public health effect impacts are highest in the manufacturing life-cycle stage for both the CRT (56% of impacts) and the LCD (65% of impacts). The use stage is the next largest contributor for the CRT (22%), and the materials processing stage in the next largest contributor for the LCD (32%). As will be shown below, use stage impacts are significant for the CRT, even when SO₂ emissions are excluded, because of the CRT's relatively high electricity consumption during use by the consumer and the associated emissions of pollutants from U.S. power plants.

Table 3-47 presents the materials that contribute greater than one percent of CRT impacts when SO_2 emissions are excluded from the CRT inventory. Under this scenario, CRT chronic

public health impacts are still being driven by emissions of criteria air pollutants,² including carbon monoxide, nitrogen oxides, and sulfur oxides (assuming that SO₂ emissions comprise a large part of the sulfur oxide emissions shown in the table). As shown in the table, emissions of these three pollutants or pollutant categories are responsible for some 48% of CRT chronic public health impacts when pure SO₂ emissions are excluded from the CRT inventory. Note that the majority of these emissions occur from the LPG production process, and most of this LPG is used as a fuel in CRT glass manufacturing. CRT glass manufacturing energy inputs are uncertain and evaluated in a sensitivity analysis (See Section 3.4). Other significant contributors include arsenic from lead production, methane from LPG production and the U.S. electric grid inventory, vanadium and benzene from LPG production, and titanium tetrachloride from aluminum production.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score
Manufacturing	LPG production	Carbon monoxide	Secondary	22.35%
Use	U.S. electric grid	Nitrogen oxides	Modeled/secondary	9.12%
Materials processing	Lead	Arsenic	Secondary	8.55%
Manufacturing	LPG production	Methane	Secondary	6.50%
Manufacturing	LPG production	Sulfur oxides	Secondary	6.21%
Use	U.S. electric grid	Methane	Modeled/secondary	4.99%
Manufacturing	LPG production	Nitrogen oxides	Secondary	4.44%
Use	U.S. electric grid	Carbon monoxide	Modeled/secondary	4.23%
Manufacturing	LPG production	Vanadium	Secondary	4.05%
Manufacturing	LPG production	Benzene	Secondary	3.32%
Materials processing	Aluminum production (virgin)	Titanium tetrachloride	Secondary	2.79%
Manufacturing	CRT glass/frit mfg.	Fluorides (F-)	Primary	2.27%
Use	U.S. electric grid	Arsenic	Modeled/secondary	2.16%
Use	U.S. electric grid	Hydrochloric acid	Modeled/secondary	1.91%
Materials processing	Steel Prod., cold-rolled, semi-finished	Carbon monoxide	Secondary	1.16%

 Table 3-47. Materials contributing greater than 1% of the CRT chronic public health effects score (without SO₂)

Table 3-48 presents the materials that contribute greater than one percent of LCD impacts when SO₂ emissions are excluded from the LCD inventory. Under this scenario, phosphine emissions from LCD monitor/module manufacturing are the dominant factor in the LCD chronic public health effects score, contributing 47% of the total. Other significant contributors include methane, benzene, carbon monoxide, and nitrogen oxides from natural gas production, and phosphorus, fluorides, tetramethyl ammonium hydroxide, and nitrogen oxides from LCD monitor/module manufacturing. Recall that the LCD monitor/module manufacturing process

² The criteria air pollutants are those for which U.S. National Ambient Air Quality Standards have been adopted. They are carbon monoxide, lead, nitrogen oxides, ozone, particulate matter, and sulfur dioxide.

consumes the majority of the natural gas made in the natural gas production process, where LNG is used as an ancillary material. However, only one of the seven LCD monitor/module manufacturers that provided inventory data to the CDP reported the ancillary use of LNG. Other LCD monitor/module manufactures did report the use of LNG as a fuel.

	public licatili circets score (without 502)				
Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score	
Manufacturing	LCD monitor/module mfg.	Phosphine	Primary	46.7%	
Materials processing	Natural gas production	Methane	Secondary	11.4%	
Materials processing	Natural gas production	Benzene	Secondary	8.61%	
Materials processing	Natural gas production	Carbon monoxide	Secondary	7.81%	
Manufacturing	LCD monitor/module mfg.	Phosphorus (yellow or white)	Primary	5.56%	
Manufacturing	LCD monitor/module mfg.	Fluorides (F-)	Primary	4.15%	
Materials processing	Natural gas production	Nitrogen oxides	Secondary	2.12%	
Manufacturing	LCD monitor/module mfg.	Tetramethyl ammonium hydroxide	Primary	2.09%	
Manufacturing	LCD monitor/module mfg.	Nitrogen oxides	Primary	1.78%	
Use	U.S. electric grid	Nitrogen oxides	Modeled/secondary	1.43%	

Table 3-48. Materials contributing greater than 1% of the LCD chronicpublic health effects score (without SO2)

3.3.13.4 Limitations and uncertainties: chronic human health effects

Most of the limitations and uncertainties in the chronic human health effects results presented here can be grouped into three categories:

- 1. *Structural or modeling limitations and uncertainties* associated with the accuracy of the toxic chemical classification method and the chemical scoring approach used to characterize human health effects.
- 2. *Toxicity data limitations and uncertainties* associated with the availability and accuracy of toxicity data to represent potential human health effects.
- 3. *LCI data limitations and uncertainties* associated with the accuracy and representativeness of the inventory data.

Each of these are discussed below.

Structural or modeling limitations and uncertainty. The chemical scoring method used in the human health effects impact characterization is a screening tool to identify chemicals of potential concern, not to predict actual effects or characterize risk. A major limitation in the method is that it only measures relative toxicity, combined with inventory amount. It does not take chemical fate, transformation, or degradation into account. In addition, it uses a simple surrogate value (i.e., inventory amount) to evaluate the potential for exposure, when actual exposure potential involves many more factors, some of which are chemical-specific. Other sources of uncertainty include possible omissions by the CDP researchers in the impact classification process (e.g., potentially toxic chemicals not classified as such) or misrepresentation of chemicals in the impact characterization method itself (e.g., misrepresenting a chemical as a small contributor to total impacts, because of missing or inaccurate toxicity data). Some of these limitations and uncertainties may also be considered limits in the toxicity data which are discussed further below.

It should also be noted, however, that because LCA involves analyzing many processes over the entire life cycle of a product, a comprehensive, quantitative risk assessment of each chemical input or output can not be done. Rather, LCA develops relative impacts that often lack temporal or spatial specificity, but can be used to identify materials for more detailed evaluation. More detailed assessments of the toxicity and potential exposures to selected materials are performed in Chapter 4.

Toxicity data limitations and uncertainties. Major uncertainties in the impact assessment for potentially toxic chemicals result from missing toxicity data and from limitations of the available toxicity data. Uncertainties in the human health hazard data (as typically encountered in a hazard assessment) include the following:

- Using dose-response data from laboratory animals to represent potential effects in humans.
- Using data from homogeneous populations of laboratory animals or healthy human populations to represent the potential effects on the general human populations, with a wide range of sensitivities.
- Using dose-response data from high dose toxicity studies to represent potential effects that may occur at low levels.
- Using data from short-term studies to represent the potential effects of long-term exposures.
- Assuming a linear dose-response relationship.
- Possibly increased or decreased toxicity resulting from chemical interactions.

Regarding uncertainties resulting from missing toxicity data, there is uncertainty associated with using a default HV (i.e., assuming average toxicity for that measure when a chemical could be either more or less toxic than average). However, the use of neutral default values for missing data reduces the bias that typically favors chemicals with little available information. Use of a data-neutral default value to fill data gaps is consistent with principles for chemical ranking and scoring (Swanson and Socha, 1997). Of the 273 chemicals classified as potentially toxic in the CDP LCA, 156 (57%) had no toxicity data for carcinogenic effects and 128 (47%) had no data for noncarcinogenic effects. Ninety-seven chemicals (36%) had no human health toxicity data whatsoever.

LCI data limitations and uncertainty. Limitations and uncertainties in the LCI data have been discussed previously and are generally related to: (1) uncertainties in data from secondary sources that may not be representative of the geographic and temporal boundaries of this LCA, and (2) uncertainties in a few of the primary data points collected specifically for this project. With regard to the latter, glass manufacturing energy inputs are particularly uncertain despite numerous attempts to resolve the uncertainty, but are responsible for a significant portion of CRT human health impacts. Glass manufacturing energy inputs are evaluated in a sensitivity analysis in Section 3.4. The amount of LNG used as an ancillary material in LCD monitor/module manufacturing is also uncertain (also despite attempts to resolve questions regarding the data), but this material contributes a significant portion of LCD occupational health impacts. As noted previously, removing this application of LNG from the LCD monitor/module manufacturing inventory would reduce the LCD chronic occupational health effects score by 68%.

3.3.13.5 Aesthetic impacts (odor)

Figure 3-20 presents the CRT and LCD LCIA results for the aesthetic impacts (odor) category, based on the impact assessment methodology presented in Section 3.1.2.12. Complete results for the CRT and LCD are presented in Tables M-35 and M-36 in Appendix M, respectively. The life-cycle aesthetic (odor) impact result is 7.58 million m³ malodorous air per functional unit for the CRT and 5.04 million m³ malodorous air per functional unit for the LCD. As shown in the figure, this impact category indicator is dominated by air emissions in the manufacturing stage for both the CRT (96% of total) and the LCD (98% of total). Both monitor types receive relatively minor contributions in the use and materials processing life-cycle stages, and negative values at end of life. Negative values are due to the offset of electric power plant emissions from incineration with energy recovery.



Major Contributors to the CRT Aesthetics (Odor) Result

Table 3-49 lists the materials that contribute to the top 99% of the CRT aesthetic impacts result and the LCI data type. Air emissions of hydrogen sulfide from LPG production in the manufacturing life-cycle stage dominate the CRT odor impacts, contributing 94% of the total score. Hydrogen sulfide impacts are calculated based on an odor threshold value (OTV) of 0.00043 mg/m³. [See Table K-7 in Appendix K for a list of OTVs used to calculate aesthetic (odor) impacts.] As noted previously, most of the LPG produced by this process is used as a fuel in CRT glass manufacturing, but glass manufacturing energy inputs are uncertain. The next largest contributor to the CRT is acetaldehyde emitted from the U.S. electric grid during the use stage. Acetaldehyde has a lower OTV (0.00027 mg/m³) than LPG, and is also emitted in smaller quantities. Emissions of hydrogen sulfide from fuel oil #6 production, steel production, and ABS production are the remaining top contributors to the CRT aesthetic impacts score. LCI data

for all of the top contributors are either from secondary data sets or developed by CDP researchers from secondary sources.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Manufacturing	LPG production	Hydrogen sulfide	Secondary	94%
Use	U.S. electric grid	Acetaldehyde	Model/secondary	2.5%
Manufacturing	Fuel oil #6 production	Hydrogen sulfide	Secondary	0.98%
Materials processing	Steel production, cold- rolled, semi-finished	Hydrogen sulfide	Secondary	0.42%
Materials processing	ABS production	Hydrogen sulfide	Secondary	0.31%

 Table 3-49. Top 99% of the CRT aesthetic (odor) impacts score

*Column may not add to 99% due to rounding.

Major Contributors to the LCD Aesthetics (Odor) Result

Table 3-50 presents the materials that contribute to the top 99% of the LCD aesthetics impact score and the LCI data type. LCD impacts are dominated by air emissions of phosphine from LCD monitor/module manufacturing, which contribute 89% of the total score. OTVs reported for phosphine range from 0.014 to 2.8 mg/m³. The lower, more sensitive value (0.014 mg/m³) was used to calculate impacts.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Manufacturing	LCD monitor/module mfg.	Phosphine	Primary	89%
Manufacturing	LPG production	Hydrogen sulfide	Secondary	6.8%
Use	U.S. electric grid	Acetaldehyde	Model/secondary	1.4%
Manufacturing	LCD monitor/module mfg.	Ammonia	Primary	1.2%
Manufacturing	LCD monitor/module mfg.	Acetic acid	Primary	0.44%

 Table 3-50.
 Top 99% of the LCD aesthetic (odor) impacts score

*Column may not add to 99% due to rounding.

Other significant contributors include hydrogen sulfide from LPG production, acetaldehyde from the U.S. electric grid, and ammonia and acetic acid from LCD module/monitor manufacturing. Most of the LPG made in the LPG production process is used as a fuel in LCD glass manufacturing, indicating this process is ultimately responsible for LPG production impacts. However, LCD glass energy inputs are uncertain and evaluated in a sensitivity analysis (Section 3.4). LCD monitor/module manufacturing data were collected directly from manufacturers by the CDP, while the LPG production inventory was obtained from *Ecobilan*. The U.S. electric grid inventory was developed by CDP researchers from secondary sources.

Limitations and Uncertainties

Aesthetic (odor) impact scores are based on the identity and amount of odor-causing chemicals (Heijungs *et al.*, 1992; EPA, 1992), released to the air divided by their chemical-specific OTVs. An OTV is the lowest concentration of a substance in air that can be smelled based on a standardized test. Limitations and uncertainties in the aesthetics impact score stem from structural or model uncertainty (whether or not odor thresholds will actually be exceeded), OTV data uncertainty (how well published OTVs represent the odor threshold of different populations), and LCI data uncertainty.

The aesthetics impact score calculates the mass of malodorous air that could result if a chemical release occurs in a finite volume of air. It does not predict whether actual odor impacts will occur. This is because LCI data do not describe the time rate of release or whether dilution and mixing with ambient air will dilute the concentration of a pollutant to below its odor threshold. In addition, odor thresholds are highly variable because of the differing ability of individuals to detect odors. Therefore, the impact scores may not account for odors perceived by the most sensitive populations or may overstate impacts perceived by less sensitive populations. Finally, the aesthetic impact scores are subject to the limitations and uncertainties in the LCI data, since they are calculated from air emissions data in the inventories. The limitations and uncertainties in LCI data were discussed in Section 2.2.2.2, and have been discussed extensively with LCIA results for other impact categories, above.

3.3.14 Ecotoxicity

Ecotoxicity refers to effects of chemical outputs on non-human living organisms. As discussed in Section 3.1.2.13, ecotoxicity impact categories included in the scope of this LCA include impacts to aquatic and terrestrial organisms.

Ecotoxicity impacts are calculated using the scoring of inherent properties approach where an impact score is based on the inventory amount weighed by a hazard value (HV). The HV represents the toxicity of a specific material to aquatic or terrestrial organisms (see Table K-8 in Appendix K for a list of toxicity values used to calculate hazard values). Aquatic HVs are based on acute and chronic toxicity values for fish, while terrestrial HVs are based on chronic noncancer toxicity values for mammals, usually rodents. Similar to the chronic human health impacts discussed in Section 3.3.13, the inventory amount (the toxic chemical outputs to water for aquatic toxicity effects, and the outputs to air and water for terrestrial toxicity effects) is used as a surrogate for exposure, while the hazard value represents the inherent toxicity of the substance.

Also like the human health effects methodology, the CDP ecotoxicity LCIA methodology does not consider the fate and transport of a toxic chemical in the environment, nor does it evaluate the potential for actual exposures to occur. In addition, the methodology is limited in that it does not consider toxicity data from all types of aquatic or terrestrial species, but rather focuses on a few selected species for which more toxicity data are available. The limitations and uncertainties in the ecotoxicity scores are discussed further below, following the presentation of results.

3.3.14.1 Aquatic toxicity

Figure 3-21 presents the CRT and LCD LCIA results for the aquatic toxicity impact category, based on the impact assessment methodology presented in Section 3.1.2.13. Complete results for the CRT and LCD are presented in Tables M-37 and M-38 in Appendix M, respectively.



The life-cycle aquatic toxicity indicator is 0.22 tox-kg per functional unit for the CRT and 5.19 tox-kg per functional unit for the LCD. As shown in the figure, the CRT aquatic toxicity indicator is driven by water releases in the materials processing stage (64% of total), while the LCD aquatic toxicity indicator is completely dominated by water releases in the manufacturing stage (99% of total). Both monitor types receive zero scores in the use stage and small, negative values at end of life. Negative values are due to the offset of electric power plant emissions from incineration with energy recovery.

Table 3-51 lists the materials that contribute to the top 99% of the CRT aquatic toxicity impact score and the LCI data type. As shown in the table, CRT aquatic toxicity impacts are broadly distributed across a number of different process groups, with most of the top contributors responsible for less than five percent of the impacts. Most of the LCI data from which the scores were calculated are from secondary data sets, although a substantial fraction are from primary data collected to meet the goals and scope of the CDP LCA. Water releases of phosphorus from CRT tube manufacturing represents the single largest contributor to the CRT aquatic toxicity score, accounting for 26% of the total. Aquatic toxicity impacts for phosphorus are driven more by its inherent acute toxicity than the output amount. The phosphorus acute HV is calculated from a fish LC_{50} of 0.020 mg/L, which is significantly more toxic than the geometric mean value of 23.5 mg/L.

The only other specific outputs that contribute more than five percent of the CRT aquatic toxicity score are water discharges of aluminum ions (valence = +3) and copper ions (valence = +1 and +2) from aluminum production. The aluminum HV is calculated from an LC_{50} value of 36 mg/L and a NOAEL value of 3.6 mg/L, which are within an order of magnitude of the geometric mean values of 23.5 mg/L and 3.9 mg/L. Copper, on the other hand, is much more toxic to fish, with an LC_{50} value of 0.014mg/L and a NOAEL value of 0.004 mg/L. The

aluminum aquatic toxicity score exceeds that of copper because it is discharged in much greater quantities.

Life-cycle stage	Process group	Material	LCI data	Contribution
			type	score*
Manufacturing	CRT tube manufacturing	Phosphorus (yellow or white)	Primary	26%
Materials processing	Aluminum production	Aluminum (+3)	Secondary	12%
Materials processing	Aluminum production	Copper (+1 & +2)	Secondary	9.5%
Materials processing	Invar	Copper (+1 & +2)	Secondary	5.0%
Materials processing	Invar	Aluminum (+3)	Secondary	4.4%
Materials processing	Invar	Zinc (+2)	Secondary	4.0%
Materials processing	Lead	Aluminum (+3)	Secondary	3.6%
Manufacturing	CRT tube manufacturing	Fluoride	Primary	3.1%
Materials processing	Ferrite manufacturing	Zinc (+2)	Secondary	3.0%
Materials processing	Aluminum production	Zinc (+2)	Secondary	2.9%
Materials processing	ABS production	Ammonia	Secondary	2.7%
Materials processing	Lead	Copper (+1 & +2)	Secondary	2.7%
Manufacturing	CRT glass/frit mfg.	Fluorides (F-)	Primary	2.6%
Manufacturing	CRT tube manufacturing	Zinc (elemental)	Primary	2.3%
Manufacturing	CRT tube manufacturing	Copper	Primary	2.1%
Materials processing	Steel production, cold- rolled, semi-finished	Phosphorus (yellow or white)	Secondary	2.0%
Manufacturing	LPG production	Phenol	Secondary	1.9%
Materials processing	Steel production, cold- rolled, semi-finished	Ammonia	Secondary	1.2%
Manufacturing	LPG production	Aluminum (+3)	Secondary	1.1%
Materials processing	Lead	Zinc (+2)	Secondary	0.82%
Materials processing	Polycarbonate production	Copper (+1 & +2)	Secondary	0.54%
Materials processing	Steel production, cold- rolled, semi-finished	Copper (+1 & +2)	Secondary	0.45%
Materials processing	Ferrite manufacturing	Aluminum (+3)	Secondary	0.43%
Materials processing	Aluminum production	Barium sulfate	Secondary	0.40%
Materials processing	ABS production	Aluminum (+3)	Secondary	0.39%
Materials processing	Invar	Ammonia	Secondary	0.36%
Materials processing	Ferrite manufacturing	Copper (+1 & +2)	Secondary	0.31%
Materials processing	ABS production	Copper (+1 & +2)	Secondary	0.25%
Materials processing	Styrene-butadiene copolymer production	Copper (+1 & +2)	Secondary	0.24%
Materials processing	Aluminum production	Titanium tetrachloride	Secondary	0.20%
Materials processing	Polycarbonate production	Mercury compounds	Secondary	0.19%
Materials processing	Aluminum production	Strontium (Sr II)	Secondary	0.14%

 Table 3-51. Top 99% of the CRT aquatic toxicity impact score

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Materials processing	Steel production, cold- rolled, semi-finished	Aluminum (+3)	Secondary	0.12%
Materials processing	Ferrite manufacturing	Ammonia	Secondary	0.12%
Materials processing	Lead	Barium sulfate	Secondary	0.10%
Materials processing	Steel production, cold- rolled, semi-finished	Nitrogen dixode	Secondary	0.088%
Materials processing	ABS production	Mercury compounds	Secondary	0.088%
Materials processing	Steel production, cold- rolled, semi-finished	Zinc (+2)	Secondary	0.087%
Materials processing	Styrene-butadiene copolymer production	Mercury compounds	Secondary	0.086%
Materials processing	Steel production, cold- rolled, semi-finished	Fluorides (F-)	Secondary	0.086%
Materials processing	Aluminum production	Lead compounds	Secondary	0.076%
Materials processing	Polycarbonate production	Zinc (+2)	Secondary	0.076%
Materials processing	Invar	Strontium (Sr II)	Secondary	0.074%

 Table 3-51. Top 99% of the CRT aquatic toxicity impact score

*Column may not add to 99% due to rounding.

Table 3-52 lists the materials that contribute to the top 99% of the LCD aquatic toxicity impact score and the LCI data type. Unlike the CRT, LCD impacts in this category are not distributed across a number of different process groups, but dominated by phosphorus emissions from a single process group, LCD monitor/module manufacturing. Phosphorus releases from LCD monitor/module manufacturing are several orders of magnitude higher than phosphorus releases from CRT tube manufacturing (the greatest contributor to the CRT aquatic toxicity impact score). However, the LCD aquatic toxicity score for phosphorus is still driven by the inherent acute toxicity of phosphorus, rather than the release amount.

Other top contributors to LCD impacts in this category include ammonia releases from PMMA sheet production, and phosphorus emissions from LCD panel components manufacturing. No toxicity data were available for ammonia. Consequently, it was assigned a default HV of two, representative of mean acute and chronic fish toxicity values.

lab	le 3-52. Top 99% of the	ELCD aquatic toxicity imj	pact score	
900	Process group	Matarial	I CI data	6

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score
Manufacturing	LCD monitor/module mfg.	Phosphorus (yellow or white)	Primary	98%
Materials processing	PMMA sheet production	Ammonia	Secondary	0.63%
Manufacturing	LCD panel components	Phosphorus (yellow or white)	Primary	0.56%

3.3.14.2 Terrestrial ecotoxicity

Figure 3-22 presents the CRT and LCD LCIA results for the terrestrial toxicity impact category, based on the impact assessment methodology presented in Section 3.1.2.13. Complete results for the CRT and LCD are presented in Tables M-39 and M-40 in Appendix M, respectively.



The life-cycle terrestrial toxicity indicator is 1,970 tox-kg per functional unit for the CRT and 894 tox-kg per functional unit for the LCD. As shown in the figure, the CRT result is dominated by toxic chemical outputs from electricity generation in the use stage, which account for almost 84% of CRT impacts in this category. To a lesser degree, LCD terrestrial toxicity impacts are also driven by emissions from electricity generation in the use stage, which account for more than 69% of the total.

The materials processing stage contributes about 12% of CRT terrestrial toxicity impacts and about five percent of LCD impacts. The manufacturing life-cycle stage is responsible for five and 26% of CRT and LCD impacts in this category, respectively. Both monitors receive very small terrestrial toxicity scores at end of life. This is because most terrestrial toxicity impacts from CRT and LCD recycling and disposal processes are offset by a credit on electric grid emissions when the monitors are incinerated with energy recovery.

The terrestrial toxicity impact results are almost identical to the chronic public health effects results presented previously (see Section 3.3.13). Recall that human health and terrestrial toxicity impacts are calculated using the same noncancer toxicity values (and the same inventory data), with the main difference being that toxicity data on carcinogenic effects are excluded from the terrestrial toxicity impact calculations. However, human health and terrestrial toxicity impacts are almost identical because: (1) impacts in both categories are dominated by emissions of sulfur dioxide from electricity generation (see Tables 3-53 and 3-54 below for top contributors to the CRT and LCD terrestrial toxicity impacts), and (2) sulfur dioxide has a high hazard value for noncancer effects and a hazard value of zero for cancer effects.

Table 3-53 presents the materials that contribute to the top 99% of the CRT terrestrial toxicity impact score. As already noted, SO₂ emissions from a number of different process groups almost completely dominate CRT impacts in this category, accounting for slightly less than 99% of the total. Most of these emissions are from the combustion of fossil fuels to generate electricity. All of the SO₂ LCI data are either from secondary data sets not developed specifically for the CDP or from the electric grid inventories developed from secondary sources for this project. Sulfur dioxide has a relatively high HV, based on an inhalation NOAEL of 0.104 mg/m³ and the geometric mean inhalation NOAEL of 68.7 mg/m³. In addition, as noted in the section on human health effects (3.3.13), from a mass loading perspective (i.e., based on the inventory alone), SO₂ emissions were the second largest contributor to CRT life-cycle air pollutant emissions, exceeded only by emissions of carbon dioxide (CO₂). Carbon dioxide is not classified as toxic, and therefore did not contribute to the terrestrial toxicity impact category.

Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score*
Use	U.S. electric grid	Sulfur dioxide	Model/secondary	83%
Materials processing	Invar	Sulfur dioxide	Secondary	8.4%
Manufacturing	Japanese electric grid	Sulfur dioxide	Model/secondary	2.9%
Manufacturing	U.S. electric grid	Sulfur dioxide	Model/secondary	1.3%
Materials processing	Steel production, cold- rolled, semi-finished	Sulfur dioxide	Secondary	1.3%
Materials processing	Aluminum production	Sulfur dioxide	Secondary	0.70%
Materials processing	Polycarbonate production	Sulfur dioxide	Secondary	0.40%
Manufacturing	LPG production	Carbon monoxide	Secondary	0.27%

 Table 3-53. Top 99% of the CRT terrestrial toxicity impact score

* Column may not add to 99% due to rounding.

Table 3-54 lists the materials that contribute to the top 99% of the LCD terrestrial toxicity impact score and the LCI data type. LCD impacts in this category are also dominated by SO₂ emissions, which are responsible for roughly 92% of impacts. Like the CRT, most of these emissions occur from electricity generation, either during the use stage (68%) or manufacturing (21%). As noted previously, SO₂ has a relatively high HV, due to its low toxicity value (inhalation NOAEL = 0.104 mg/m³). Similar to the CRT, from a mass loading perspective (i.e., based on the inventory alone), SO₂ emissions were the second largest contributor to LCD lifecycle air pollutant emissions, exceeded only by emissions of CO₂.

Other top contributors to the LCD terrestrial toxicity score include phosphine and phosphorus from LCD monitor/module manufacturing, and carbon monoxide and methane from natural gas production. As noted above in the section on chronic occupational health effects, the phosphine, phosphorus, and benzene scores are driven more by their inherent toxicity than their output amounts.

Tuble e on Top >> /o of the Lob terrestrial tokienty impact score					
Life-cycle stage	Process group	Material	LCI data type	Contribution to impact score	
Use	U.S. electric grid	Sulfur dioxide	Model/secondary	68%	
Manufacturing	Japanese electric grid	Sulfur dioxide	Model/secondary	21%	
Manufacturing	LCD monitor/module mfg.	Phosphine	Primary	3.2%	
Materials processing	Steel production, cold- rolled, semi-finished	Sulfur dioxide	Secondary	1.4%	
Materials processing	PMMA sheet production	Sulfur dioxide	Secondary	0.96%	
Materials processing	Natural gas production	Benzene	Secondary	0.59%	
Materials processing	Aluminum production	Sulfur dioxide	Secondary	0.57%	
Materials processing	Natural gas production	Carbon monoxide	Secondary	0.50%	
Materials processing	Polycarbonate production	Sulfur dioxide	Secondary	0.50%	
Materials processing	Natural gas production	Methane	Secondary	0.39%	
Manufacturing	LCD monitor/module mfg.	Phosphorus (yellow or white)	Primary	0.38%	

 Table 3-54. Top 99% of the LCD terrestrial toxicity impact score

3.3.14.3 Terrestrial toxicity impact scores modified to exclude sulfur dioxide

Because the terrestrial toxicity impact scores for both the CRT and the LCD are dominated by SO_2 emissions, a secondary analysis was run to identify the top contributors to these impacts when SO_2 emissions are excluded from the inventories. Results of this analysis may be more useful to manufacturers seeking to identify problematic toxic chemicals within their own manufacturing processes.

Figure 3-23 presents the CRT and LCD terrestrial toxicity impact scores by life-cycle stage when SO₂ emissions are excluded from the inventories. Under this scenario, the CRT score is reduced almost 99% from 1,970 tox-kg to 21 tox-kg per functional unit, and the LCD score is reduced about 94%, from 894 tox-kg to 54 tox-kg per functional unit. Note that these scores should not be used to evaluate which monitor type has higher overall impacts in this category, but they are useful for identifying life-cycle improvement opportunities that were previously obscured by SO₂ impacts.



With SO_2 emissions removed from the inventories, terrestrial toxicity impacts are highest in the manufacturing life-cycle stage for both the CRT (56% of impacts) and the LCD (70% of impacts). The materials processing stage is the next largest contributor for both monitor types, contributing 25% of CRT impacts and 27% of LCD impacts, followed by the use stage, which contributes 20% of CRT impacts and almost 3% of LCD impacts. Both monitors have slight negative values at end-of-life, due to the offset of electric grid emissions from incineration with energy recovery.

Table 3-55 presents the materials that contribute greater than one percent of CRT terrestrial toxicity impacts when SO₂ emissions are excluded from the CRT inventory. Like the modified public chronic human health effects results discussed in Section 3.3.13, under this scenario CRT chronic public health impacts are still being driven by emissions of criteria air pollutants, including carbon monoxide, nitrogen oxides, and sulfur oxides (assuming that SO₂ emissions comprise a large part of the sulfur oxide emissions shown in the table). As shown in the table, emissions of these three pollutants or pollutant categories are responsible for some 46% of CRT terrestrial toxicity impacts when pure SO₂ emissions are excluded from the CRT inventory. Note that the majority of these emissions occur from the LPG production process, and most of this LPG is used as a fuel in CRT glass manufacturing. CRT glass manufacturing energy inputs are uncertain and evaluated in a sensitivity analysis (see Section 3.4). Other significant contributors include arsenic from lead production, methane from LPG production and the U.S. electric grid inventory, vanadium and benzene from LPG production, and titanium tetrachloride from aluminum production.

I fo analo stage					
Life-cycle stage	Process group	Material	LCI data	Contribution	
			type	to impact	
				score	
Manufacturing	LPG production	Carbon monoxide	Secondary	26%	
Materials processing	Lead	Arsenic	Secondary	11%	
Use	U.S. electric grid	Nitrogen oxides	Model/secondary	5.7%	
Manufacturing	LPG production	Vanadium	Secondary	5.1%	
Use	U.S. electric grid	Carbon monoxide	Model/secondary	4.9%	
Manufacturing	LPG Production	Benzene	Secondary	4.2%	
Manufacturing	LPG production	Methane	Secondary	4.1%	
Manufacturing	LPG production	Sulfur oxides	Secondary	3.9%	
Materials processing	Aluminum production	Titanium tetrachloride	Secondary	3.5%	
	(all virgin)				
Use	U.S. electric grid	Methane	Model/secondary	3.1%	
Manufacturing	CRT glass/frit mfg.	Fluorides (F-)	Primary	2.8%	
Manufacturing	LPG production	Nitrogen oxides	Secondary	2.8%	
Use	U.S. electric grid	Arsenic	Model/secondary	2.7%	
Use	U.S. electric grid	Hydrochloric acid	Model/secondary	2.4%	
Materials processing	Steel Prod., cold-rolled,	Carbon monoxide	Secondary	1.4%	
	semi-finished				
Materials processing	Invar	Titanium tetrachloride	Secondary	1.2%	
Manufacturing	CRT tube manufacturing	Carbon monoxide	Primary	1.0%	
Materials processing	Lead	Titanium tetrachloride	Secondary	1.0%	
Manufacturing	LPG production	Arsenic	Secondary	1.0%	

Table 3-55.	Materials cont	tributing greate	er than 1% of the	CRT
te	rrestrial toxicit	y impact score	(without SO ₂)	

Table 3-56 presents the materials that contribute greater than one percent of LCD terrestrial toxicity impacts when SO₂ emissions are excluded from the LCD inventory. As with the LCD modified chronic human health results discussed in Section 3.3.13, when SO₂ emissions are excluded, phosphine emissions from LCD monitor/module manufacturing are the dominant factor in the LCD terrestrial toxicity score, contributing 53% of the total. Other significant contributors include benzene, carbon monoxide, methane, and nitrogen oxides from natural gas production, and phosphorus, fluorides, tetramethyl ammonium hydroxide, and nitrogen oxides from LCD monitor/module manufacturing process consumes the majority of the natural gas made in the natural gas production process, where LNG is used as an ancillary material. However, only one of the seven LCD monitor/module manufacturers that provided inventory data to the CDP reported the ancillary use of LNG. Other manufacturers reported using LNG as a fuel.

terrestriar toxicity impact score (without soy)						
Life-cycle stage	Process group	group Material		Contribution to impact score		
Manufacturing	LCD module/monitor mfg.	Phosphine	Primary	53%		
Materials processing	Natural gas production	Benzene	Secondary	9.8%		
Materials processing	Natural gas production	Carbon monoxide	Secondary	8.2%		
Materials processing	Natural gas production	Methane	Secondary	6.5%		
Manufacturing	LCD module/monitor mfg.	Phosphorus (yellow or white)	Primary	6.3%		
Manufacturing	LCD module/monitor mfg.	Fluorides (F-)	Primary	4.7%		
Materials processing	Natural gas production	Nitrogen oxides	Secondary	1.2%		
Manufacturing	LCD module/monitor mfg.	Tetramethyl ammonium hydroxide	Primary	1.2%		
Manufacturing	LCD module/monitor mfg.	Nitrogen oxides	Primary	1.0%		

Table 3-56. Materials contributing greater than 1% of the LCDterrestrial toxicity impact score (without SO2)

3.3.14.4 Limitations and uncertainties

Most of the limitations and uncertainties in the ecotoxicity results are similar to the limitations and uncertainties in the human health effects scores. The reader is referred to Section 3.3.13 for a full discussion of these limitation and uncertainties. In summary, they can be grouped into three categories:

- 1. *Structural or modeling limitations and uncertainties* associated with the accuracy of the toxic chemical classification method and the chemical scoring approach used to characterize human health effects.
- 2. *Toxicity data limitations and uncertainties* associated with the availability and accuracy of toxicity data to represent ecotoxicity.
- 3. *LCI data limitations and uncertainties* associated with the accuracy and representativeness of the inventory data.

With regard to toxicity data, other limitations and uncertainties in the ecotoxicity results are related to the use of surrogates to assess toxicity to all species within an impact category. For example, the aquatic toxicity category uses fish (usually the fathead minnow) as a surrogate to assess toxicity to all aquatic organisms, but it has been well-established in the ecotoxicology literature that fish are not the most sensitive test species to all or most industrial chemicals. In fact, invertebrates (daphnids) or algae (Selenastrum) often are more sensitive to particular chemicals than fish (Smrchek, 1999). Similarly, the terrestrial toxicity category uses mammals, primarily rodents, as a surrogate to assess toxicity to all terrestrial organisms. Terrestrial plants and soil organisms (insects, earthworms, etc.) are not considered, but both of these may be more sensitive than mammals.

Because there are difficulties in comparing test endpoints for different types of organisms, and because there is a very limited toxicity database for some of the other organisms, the LCIA methodology employed in this study uses fish as a surrogate for aquatic toxicity and mammals as a surrogate for terrestrial ecotoxicity. This helps to reduce data gaps and the difficulties in comparing test endpoints for different types of organisms. Furthermore, we believe this approach to be acceptable for a study, such as the CDP LCA, that gives relative

ranking of impacts from different chemicals or process groups instead of absolute values. However, it should be noted that this approach can result in an underestimation of the absolute ecotoxicity and hazards of chemicals.

With regard to LCI data limitations not discussed previously, it should be noted that the CRT and LCD LCAs do not address spills or other accidental releases that could have significant adverse effects on aquatic or terrestrial organisms. This is a common limitation of LCA, which is often too labor-intensive to address different operational scenarios across the product life cycle.

3.3.15 Summary of Top Contributors by Impact Category

Tables 3-57 and 3-58 summarize the top contributors to CRT and LCD life-cycle impacts by impact category. As shown in Table 3-57, CRT impacts are largely driven by two factors: (1) the large amount of LPG fuel used in CRT glass/frit manufacturing, and (2) the relatively large amount of electricity consumed during the use stage. The LPG production process yields the CRT's top contributor in eight of 20 impact categories. Most of this LPG is used as a fuel source in CRT glass manufacturing in the glass/frit process group, which, in turn, produces the top contributor to two of 20 impact categories. Thus, LPG used in the glass/frit process group (primarily CRT glass manufacturing) is ultimately the key driver for CRT impacts in ten categories. Similarly, outputs from electricity generation during the use stage result in the top contributor to seven CRT impact categories. Note that in 14 of the 20 impact categories, the top contributor to CRT impacts is responsible for more than 50% of impacts.

Both the glass manufacturing energy and the use stage lifespan (which determines the amount of electricity generated during the use stage) are evaluated in a sensitivity analysis in Section 3.4. In the modified glass energy sensitivity analysis, LPG inputs are greatly reduced and impacts are therefore reduced, but in the modified lifespan (manufactured life) sensitivity analysis, the number of hours a monitor is in use is increased. Thus, CRT impacts are increased.

LCD impacts are not as dominated by a few data points, but a few processes (LCD monitor/module manufacturing and electricity generation in the use stage) are responsible for a large percent of the impacts. As shown in Table 3-58, both of these processes result in the top contributors to six LCD impact categories each. In addition, the process to produce LNG used as an ancillary material in LCD monitor/module manufacturing is the top contributor to an additional impact category (photochemical smog). Note that in 11 of the 20 impact categories, the top contributor to LCD impacts is responsible for more than 50% of impacts.

Like the CRT, both the glass energy inputs and use stage lifespan of the LCD are evaluated in a sensitivity analysis in Section 3.4. LCD monitor/module manufacturing energy and LCD EOL dispositions are also evaluated. LCD monitor/module manufacturing energy was selected for a sensitivity analysis because of the high degree of variability seen in data provided by manufacturers. The impacts presented in the baseline scenario are calculated with energy outliers removed from the average, but the outliers are included in the sensitivity analysis. This results in higher electricity consumption but lower fuel consumption, which, in turn, causes reduced impacts in some categories and increased impacts in others. Note that the LNG used as an ancillary material in LCD monitor/module manufacturing is not affected by the sensitivity analysis since it only focuses on materials used as an energy source.

Impact category	Top contributors			
	Life-cycle stage	Process group	Material	Contribution to impact score
Renewable resource use	Manufacturing	LPG production	water	79%
Nonrenewable resource use	Manufacturing	LPG production	Petroleum (in ground)	56%
Energy use	Manufacturing	CRT glass/frit mfg.	Liquefied petroleum gas	72%
Solid waste landfill use	Use	U.S. electric grid	Coal waste	38%
Hazardous waste landfill use	End-of-life	CRT landfilling	EOL CRT monitor, landfilled	91%
Radioactive waste landfill use	Use	U.S. electric grid	Low-level radioactive waste	61%
Global warming	Use	U.S. electric grid	Carbon dioxide	64%
Ozone depletion	Use	U.S. electric grid	Bromomethane	49%
Photochemical smog	Manufacturing	LPG production	Hydrocarbons, unspeciated	36%
Acidification	Use	U.S. electric grid	Sulfur dioxide	47%
Air particulates	Manufacturing	LPG production	PM	43%
Water eutrophication	Manufacturing	LPG production	COD	72%
Water quality, BOD	Manufacturing	LPG production	BOD	96%
Water quality, TSS	Manufacturing	LPG production	Suspended solids	97%
Radioactivity	Materials Processing	Steel production, cold- rolled, semi-finished	Plutonium-241 (isotope)	62%
Chronic health effects, occupational	Manufacturing	CRT glass/frit manufacturing	Liquefied petroleum gas	78%
Chronic health effects, public	Use	U.S. electric grid	Sulfur dioxide	83%
Aesthetics (odor)	Manufacturing	LPG production	Hydrogen sulfide	94%
Aquatic toxicity	Manufacturing	CRT tube manufacturing	Phosphorus (yellow or white)	26%
Terrestrial toxicity	Use	U.S. electric grid	Sulfur dioxide	83%

 Table 3-57. Summary of top contributors to CRT impacts by impact category
Impact category	Top contributors					
	Life-cycle stage	Process group	Material	Contribution to impact score		
Renewable resource use	Manufacturing	LCD monitor/module mfg.	Water	38%		
Nonrenewable resource use	Materials processing	Natural gas production	Natural gas (in ground)	65%		
Energy use	Use	LCD monitor use	Electricity	30%		
Solid waste landfill use	Use	U.S. electric grid	Coal waste	44%		
Hazardous waste landfill use	End-of-life	LCD landfilling	EOL LCD monitor, landfilled	97%		
Radioactive waste landfill use	Use	U.S. electric grid	Low-level radioactive waste	44%		
Global warming	Manufacturing	LCD monitor/module mfg.	Sulfur hexafluoride	29%		
Ozone depletion	Manufacturing	LCD panel components manufacturing	HCFC-225cb	34%		
Photochemical smog	Materials processing	Natural gas production	Nonmethane hydrocarbons, unspeciated	45%		
Acidification	Use	U.S. electric grid	Sulfur dioxide	31%		
Air particulates	Materials processing	Steel production, cold- rolled, semi-finished	РМ	45%		
Water eutrophication	Manufacturing	LCD monitor/module mfg.	Nitrogen	67%		
Water quality, BOD	Manufacturing	LCD monitor/module mfg.	BOD	61%		
Water quality, TSS	Manufacturing	LPG production	Suspended solids	66%		
Radioactivity	Materials processing	Steel production, cold- rolled, semi-finished	Plutonium-241 (isotope)	96%		
Chronic health effects, occupational	Manufacturing	LCD monitor/module mfg.	Liquefied natural gas	58%		
Chronic health effects, public	Use	U.S. electric grid	Sulfur dioxide	68%		
Aesthetics (odor)	Manufacturing	LPG production	Hydrogen sulfide	94%		
Aquatic toxicity	Manufacturing	LCD monitor/module mfg.	Phosphorus (yellow or white)	98%		
Terrestrial toxicity	Use	U.S. electric grid	Sulfur dioxide	68%		

 Table 3-58.
 Summary of top contributors to LCD impacts by impact category

3.4 SENSITIVITY ANALYSES

Due to assumptions and uncertainties in this LCA, as in any LCA, several sensitivity analyses of the baseline results were conducted. Section 2.7.3 described how areas for sensitivity analyses (scenarios) were selected, and the modifications made to the baseline inventory. Sections 3.4.1 through 3.4.4 recap these modifications and present sensitivity analysis results. Section 3.4.5 summarizes the effects of different scenarios on CRT and LCD impacts.

3.4.1 Manufactured Life Scenario

Due to the uncertainty and assumptions associated with the baseline use stage lifespan (effective life) scenario, a "manufactured life" scenario was also considered. Recall that the manufactured life is defined as the length of time a monitor is designed to operate effectively, while the effective life is defined as the actual amount of time a monitor is used, by one or multiple users, before it reaches its final disposition. The manufactured life is the number of hours a monitor would function as manufactured, and is independent of user choices or actions. Section 2.4.1.3 presented a detailed discussion of how the manufactured life was determined, and is summarized below.

The manufactured life of both monitor types was estimated using the mean-time-before-failure (MTBF) specifications of the monitor and its components. From review of MTBF information obtained on CRT-based monitors (see Appendix H, Attachment A, Table A2), it appears that the CRT tube itself is the component that 99% of the time determines whether the entire monitor has reached its end-of-life. Thus, an average of the two ranges obtained on the estimated lifetime of CRT tubes (10,000 and 15,000 hours) was used as the CRT manufactured lifetime (12,500 hours).

For active matrix LCDs, the components that have the greatest potential to fail first are the display panel itself (including the liquid crystals and thin-film transistors), backlights, driver integrated circuit (IC) tabs, and other smaller components. The backlights and driver IC tabs can be field-replaced, thus their failure does not necessarily represent the end of the monitor's life. However, failure of the liquid crystals or transistors, which would require replacement of the display panel itself, would most likely mean that the monitor cannot be cost-effectively repaired. Thus, in this study, the amount of time an LCD monitor would operate during its manufactured life is assumed to be the average of the non field-replaceable values, or 45,000 hours. In order for a monitor to operate for 45,000 hours, any major field-replaceable parts that have MTBFs less than 45,000 hours are accounted for in the inventory. For example, assuming the backlights last on average 32,500 hours (the average of the values obtained for backlights), approximately 1.4 backlights on average would be needed for every panel during its 45,000 hour lifetime.

To calculate the manufactured life electricity consumption (kWh/life), the energy use rate (kW) was multiplied by the lifespan (hours/life) for each monitor in each power mode (see Table 2-20 in Section 2.4.1.3). The LCD manufactured life (45,000 hours) is 3.6 times greater than the CRT manufactured life (12,500 hours). In an LCA, comparisons are made based on functional equivalency. Therefore, if one monitor will operate for a longer period of time than another, impacts should be based on an equivalent use. Thus, based on equivalent use periods, 3.6 CRTs

would need to be manufactured for every LCD. This was incorporated into the profile analysis for the manufactured life LCA. To apply the manufactured life scenario to the CRT and LCD

life-cycle profiles, the following modifications were made to the baseline (effective life) scenario:

- change the CRT electricity input in the use stage from 635 kWh (2,286 MJ) to 788 kWh (2,837 MJ); ³
- change the LCD electricity input in the use stage from 237 kWh (853 MJ) to 1,035 kWh (3,726 MJ);
- increase the manufacturing of CRTs by a factor of 3.6 to account for the functional equivalency of CRTs and LCDs. This was done by increasing the functional unit by a factor of 3.6, which equates to manufacturing, using, and recycling or disposing of 3.6 times more CRTs than in the baseline case; and
- increase the manufacturing of the LCD backlight lamp by a factor of 1.4 to account for the functional equivalency of LCDs and CRTs. This was done by increasing the backlight lamp mass (0.0023 kg) by a factor of 1.4, which in turn results in an increase in inputs and outputs associated with manufacturing the backlight.

Table 3-59 presents the CRT and LCD life-cycle results by impact category for the baseline and manufactured life scenarios. It also presents the percent change from the baseline to the manufactured life scenario for both monitor types. Note that the manufactured life results are most useful for evaluating the CRT and LCD together, not for comparing the CRT or LCD baseline (effective life) results to its manufactured life results. This is because the CRT and LCD to LCD effective life and manufactured life scenarios are not. The baseline for both monitor types represents impacts from manufacture and final disposition of one monitor. The CRT manufactured life scenario represents impacts from 3.6 CRT monitors and the LCD manufactured life scenario represents impacts from one LCD monitor and 1.4 backlights. However, the percent change figures are presented to better understand how the manufactured life scenario affects overall results.

 $^{^{3}}$ This represents the electricity use for a 12,500 hour life span. This figure is then multiplied by a factor of 3.6 in the functional equivalency calculations (see third bullet, below).

			CRT			LCD		
Impact category	Units/ Monitor	Baseline	Manu-	%	Baseline	Manu-	%	
			factured	change		factured	change	
Renewable resource use	kg	1.31e+04	4.83e+04	268%	2.80e+03	4.30e+03	53.5%	
Nonrenewable resource use	kg	6.68e+02	2.58e+03	286%	3.64e+02	6.12e+02	68.0%	
Energy use	MJ	2.08e+04	7.70e+04	270%	2.84e+03	5.71e+03	101%	
Solid waste landfill use	m3	1.67e-01	6.86e-01	312%	5.43e-02	1.79e-01	230%	
Hazardous waste landfill use	m3	1.68e-02	6.05e-02	260%	3.60e-03	3.60e-03	0.00%	
Radioactive waste landfill use	m3	2.00e-04	8.00e-04	328%	1.00e-04	3.00e-04	194%	
Global warming	kg-CO2 eq.s	6.95e+02	2.90e+03	317%	5.93e+02	1.17e+03	97.3%	
Ozone depletion	kg-CFC-11 eq.s	2.05e-05	8.27e-05	304%	1.37e-05	2.66e-05	94.1%	
Photochemical smog	kg-ethene eq.s	1.71e-01	6.20e-01	262%	1.41e-01	1.47e-01	4.22%	
Acidification	kg-SO2 eq.s	5.25e+00	2.19e+01	317%	2.96e+00	7.29e+00	146%	
Air particulates	kg	3.01e-01	1.13e+00	277%	1.15e-01	1.87e-01	63.4%	
Water eutrophication	kg-phosphate	4.82e-02	1.74e-01 ^a	260%	4.96e-02	4.96e-02	0.02%	
	eq.s							
BOD	kg	1.95e-01	7.02e-01	260%	2.83e-02	2.83e-02	0.02%	
TSS	kg	8.75e-01	3.15e+00	260%	6.15e-02	6.15e-02	0.02%	
Radioactivity	Bq	3.85e+07	1.14e+08	197%	1.22e+07	1.28e+07	4.49%	
Chronic health effects,	tox-kg	9.34e+02	3.39e+03	263%	6.96e+02	7.41e+02	6.47%	
occupational								
Chronic health effects, public	tox-kg	1.98e+03	8.56e+03	333%	9.02e+02	2.98e+03	230%	
Aesthetics (odor)	m3	7.58e+06	2.74e+07	262%	5.04e+06	5.30e+06	5.00%	
Aquatic toxicity	tox-kg	2.25e-01	8.10e-01	260%	5.19e+00	5.19e+00	0.02%	
Terrestrial toxicity	tox-kg	1.97e+03	8.54e+03	333%	8.94e+02	2.97e+03	232%	

Table 3-59. Baseline and sensitivity analysis results—manufactured life

^a Bold indicates impact category indicator that reversed direction from the baseline scenario such that the CRT indicator is now greater than the LCD.

As shown in Table 3-59, under the manufactured life scenario CRT impacts exceed those of the LCD in every impact category except aquatic toxicity. CRT impacts were expected to be greater than those of the LCD in most impact categories for the following reasons:

- Under the baseline scenario CRT impacts exceeded those of the LCD in every category but water eutrophication and aquatic toxicity.
- The manufactured life scenario assumes more CRTs are manufactured than LCDs during the manufactured life lifespan, which results in greater impacts.
- The manufactured life use stage is longer than the baseline, effective life use stage, and the CRT consumes more electricity during use than the LCD.

By looking at the percent change in impact scores from the baseline to manufactured life for a monitor type we can better understand which aspect of the life-cycle is driving impacts. For example, CRT impacts increased by roughly 260% in several impact categories, which is the increase from manufacturing or disposing of an additional 2.6 monitors. Energy impacts increased by more than 260% due to the additional increase in electricity consumption during use. The CRT chronic public human health effects category increased by some 330%. This is explained by the increase in SO₂ emissions in the use stage and the high HV for SO₂, which has a proportionately greater effect on overall impacts than increased outputs of other pollutants with lower HVs in other life-cycle stages.

LCD impacts increased only slightly from the baseline to the manufactured life scenario in some impact categories, but increased up to 230% in others. Most of the LCD impact categories with less than one percent increase are for impacts related to water discharges (e.g., water eutrophication, aquatic toxicity, etc.). This is because the most significant change to the LCD inventory from the baseline to the manufactured life scenario was in the use stage, and few water discharges are reported in the U.S. electric grid inventory. On the other hand, the chronic public health effects and terrestrial toxicity impact categories show the greatest increase. These results are driven by air emissions of SO₂ from U.S. power production, which increased significantly with the longer lifespan.

To further illustrate how the longer lifespan (and additional manufacturing requirements, mainly for the CRT) in the manufactured life scenario is affecting impacts, Figures 3-24 and 3-25 compare the energy impacts and public chronic health effects, respectively, of both monitor types under the baseline and manufactured life scenarios. As shown in Figure 3-24, CRT energy impacts are still dominated by the manufacturing stage in the manufactured life scenario. This is mainly due to the large amount of LPG used to manufacture 3.6 sets of CRT glass. On the other hand, LCD energy impacts during the use stage exceeded those in manufacturing by a factor of about 2.6 in the baseline scenario, but are ten times greater in the manufactured life scenario. This is due to the longer lifespan for a single LCD in the manufactured life scenario.



Figure 3-25 shows that CRT chronic public health impacts are similarly distributed in the baseline and manufactured life scenarios. However, a greater percentage of LCD chronic health impacts are in the use stage under the manufactured life scenario than the baseline due to the longer use stage.



3.4.2 Modified Glass Energy Scenario

One area of relatively large uncertainty and variability in the primary data supplied for the project was the glass manufacturing data. Both the CRT and LCD glass data were based on data supplied by CRT lead oxide (PbO) glass manufacturers, since no LCD glass manufacturers were willing to provide data to the project. To represent an LCD glass inventory, lead (Pb) materials in the CRT glass inventory were removed. Based on conversations with industry members, it was assumed that the same amount of energy per kilogram of glass produced is used to generate LCD and CRT glass. In addition, large variability among the three data sets collected resulted in a large degree of uncertainty in the glass inventory. Finally, a number of the CRT impact category results are being driven by the glass energy data or by the production process for producing the large amount of LPG used as a fuel in glass manufacturing. Consequently, the glass manufacturing inventories for both LCD and CRT glass were modified and life-cycle impacts were recalculated.

To conduct the sensitivity analysis, the energy input data for glass manufacturing were modified by removing, from the average, data that appeared unusually large, and that might be inconsistent with general industry statistics. However, industry statistics are greatly lacking when specifically considering specialty glasses such as CRT and LCD glass. Modifying the energy inputs greatly reduced the fuel energy amounts reported for the glass production process.

The baseline scenario, based on averaged primary data from manufacturers, assumed that the total energy to produce a kilogram of CRT or LCD glass was 1,560 MJ (433 kWh) of energy, with only 0.3% of that as electrical energy. The sensitivity analysis scenario assumes 16.3 MJ (4.5 kWh) per kilogram of glass produced, with approximately 30% as electrical energy. The majority of the fuel energy in the baseline scenario was from LPG. The actual input amounts are not presented here to protect the confidentiality of data provided by glass manufacturers.

Under the sensitivity (modified glass energy) scenario, only the manufacturing stage is affected, since the production of the fuels used during manufacturing is included in the manufacturing life-cycle stage. All impact categories, not only energy, are also affected because

the inputs and outputs from fuel production and electricity generation processes affect each of the impact categories evaluated in this study.

Table 3-60 shows the baseline impact results and the revised impact results based on the modified glass energy inputs. The overall life-cycle impact results are highly sensitive to the energy consumption values from glass manufacturing. Under the modified glass energy scenario, the nonrenewable resource use, global warming, photochemical smog, BOD, TSS, chronic occupational health effects, and odor impact categories reversed direction such that the LCD had greater impacts within each impact category than the CRT in the overall life cycle. Note that the percent change in CRT results in most impact categories is much greater than that of the corresponding LCD results. This is because the CRT uses approximately ten times more glass than the LCD and therefore, the CRT results are much more sensitive to the glass manufacturing data than are the LCD results.

			CRT		LCD			
Impact category	Units/Monitor	Baseline	Glass	%	Baseline	Glass	%	
			energy	change		energy ^a	change	
Renewable resource use	kg	1.31e+04	2.67e+03	-79.6%	2.80e+03	2.43e+03	-13.4%	
Nonrenewable resource use	kg	6.68e+02	2.35e+02	-64.8%	3.64e+02	3.49e+02	-4.14%	
Energy use	MJ	2.08e+04	3.02e+03	-85.5%	2.84e+03	2.04e+03	-28.0%	
Solid waste landfill use	m3	1.67e-01	1.23e-01	-26.0%	5.43e-02	5.27e-02	-2.82%	
Hazardous waste landfill use	m3	1.68e-02	1.54e-02	-8.13%	3.60e-03	3.60e-03	-1.35%	
Radioactive waste landfill use	m3	2.00e-04	2.00e-04	0.12%	1.00e-04	1.00e-04	0.01%	
Global warming	kg-CO2 eq.s	6.95e+02	5.23e+02	-24.8%	5.93e+02	5.87e+02	-1.01%	
Ozone depletion ^b	kg-CFC-11 eq.s	2.05e-05	1.97e-05	-3.75%	1.37e-05	1.37e-05	-0.18%	
Photochemical smog	kg-ethene eq.s	1.71e-01	5.59e-02	-67.3%	1.41e-01	1.37e-01	-2.84%	
Acidification	kg-SO2 eq.s	5.25e+00	4.02e+00	-23.4%	2.96e+00	2.92e+00	-1.45%	
Air particulates	kg	3.01e-01	1.72e-01	-42.9%	1.15e-01	1.10e-01	-3.97%	
Water eutrophication	kg-phosphate eq.s	4.82e-02	4.10e-03	-91.5%	4.96e-02	4.80e-02	-3.17%	
BOD	kg	1.95e-01	7.00e-03	-96.4%	2.83e-02	2.16e-02	-23.8%	
TSS	kg	8.75e-01	2.06e-02	-97.6%	6.15e-02	3.09e-02	-49.7%	
Radioactivity	Bq	3.85e+07	3.16e+07	-17.9%	1.22e+07	1.22e+07	0.00%	
Chronic health effects, public	tox-kg	1.98e+03	1.97e+03	-0.56%	9.02e+02	9.01e+02	-0.02%	
Chronic health effects,	tox-kg	9.34e+02	2.30e+02	-75.4%	6.96e+02	6.63e+02	-4.74%	
occupational	_							
Aesthetics (odor)	m3	7.58e+06	1.09e+04	-99.9%	5.04e+06	4.79e+06	-4.99%	
Aquatic toxicity	tox-kg	2.25e-01	2.18e-01	-3.07%	5.19e+00	5.19e+00	0.01%	
Terrestrial toxicity	tox-kg	1.97e+03	1.97e+03	-0.42%	8.94e+02	8.94e+02	-0.01%	

Table 3-60. Baseline and sensitivity analysis results—modified glass energy

^a Bold indicates impact category indicator that reversed direction from the baseline scenario such that the LCD indicator is now greater than the CRT.

^b LCD impacts in this category are greater than CRT impacts when phased out substances are removed from the inventories (see Section 3.3.6).

The energy impacts for the baseline and modified glass energy scenarios are presented in Figure 3-26. In the baseline scenario, over 18,000 MJ of energy were consumed per CRT monitor during manufacturing. Almost 83% of this was from the glass/frit process group, mainly from glass manufacturing energy alone. When the glass energy inputs are reduced under the modified scenario, total energy use in the CRT manufacturing stage decreases some 97% to just under 500 MJ, and the use stage dominates the overall life-cycle energy impacts at

approximately 2,300 MJ per functional unit (i.e., per monitor). The 97% decrease in manufacturing stage energy use is due to the reduced glass manufacturing fuel inputs and the consequent reduction in energy inputs to the fuel production process.



The modified glass energy scenario has a lesser, but still significant, effect on the distribution of LCD energy impacts across life-cycle stages. Under the sensitivity scenario, the LCD manufacturing stage energy consumption is reduced 55% from 1,440 MJ per monitor to about 640 MJ per monitor, and the use stage becomes the biggest energy consumer at about 850 MJ per monitor.

Global warming is one of the impact categories in which the CRT has the greater impacts than the LCD under the baseline scenario, but the LCD has the greater impacts under the sensitivity analysis. Figure 3-27 shows the global warming impacts for both monitor types under the baseline and modified glass energy scenarios. Under the latter scenario, CRT global warming impacts in the manufacturing stage are reduced some 85%, but LCD impacts are only reduced 2.5%. Again, this illustrates the greater sensitivity of CRT impact results to glass energy inputs. Also, as discussed in Section 3.3.5, a large part of LCD global warming impacts are driven by sulfur hexafluoride emissions from LCD monitor/module manufacturing, which are unaffected by the revised glass energy scenario.



3.4.3 Modified LCD Module Energy Scenario

LCD monitor/module manufacturing energy was another area of relatively large uncertainty and variability in the inventory data. As discussed in Section 2.7.3.3, total energy inputs reported in six data sets received from LCD monitor/module manufacturers in Japan and Korea ranged from 330 MJ to 7,310 MJ, with a mean and standard deviation of 2,269 MJ and 2,906 MJ, respectively. The manufacturing energy data reported in two of the data sets were found to be outliers and removed from the averages used in the baseline inventory. However, for this sensitivity analysis, the outliers were added back in to the averages. Thus, to apply the modified LCD manufacturing energy scenario to the LCD profile, the following modifications were made to electricity and fuels:

- changed the electric energy inputs to the LCD monitor/module manufacturing process group from 82.1 kWh (253 MJ) per monitor to 70.1 kWh (217 MJ) per monitor,
- changed the fuel oil # 4 inputs from 0.25 kg to 0.30 kg per monitor,
- changed the kerosene inputs from 0.35 kg to 0.23 kg per monitor,
- changed the LPG inputs from 0.68 kg to 0.45 kg per monitor,
- changed the LNG inputs from 3.8 kg to 45 kg per monitor, and
- changed the natural gas inputs from 0.99 to 0.70 kg per monitor.

Note that one LCD monitor/module manufacturer also reported using a large amount of LNG as an ancillary material. However, this input amount is not affected by the sensitivity analysis, which only deals with inputs used as an energy source.

Table 3-61 presents the baseline impact results and the revised LCD impact results based on the modified LCD module energy scenario. It also shows the baseline CRT results. Under the modified LCD energy scenario, LCD impacts in ten categories actually decrease slightly, due to the slight decrease in average electrical energy consumed during LCD monitor/module manufacturing. However, impacts in six categories increase slightly, and impacts in four categories (nonrenewable resource use, energy use, photochemical smog and chronic

3.4 SENSITIVITY ANALYSES

occupational health effects) increase by more than 10%. As expected, life-cycle energy impacts are the most affected by this sensitivity analysis, due to the increased fuel consumption during manufacturing. However, under this scenario, none of the impact category results reversed direction from the baseline such that the LCD now has greater impacts than the CRT or vice versa, where the baseline LCD impacts were greater than the CRT.

Impact category	Units/Monitor	CRT	LCD				
		Baseline	Baseline	Mod. energy	% change		
Renewable resource use	kg	1.31e+04	2.80e+03	2.78e+03	-0.69%		
Nonrenewable resource use	kg	6.68e+02	3.64e+02	4.06e+02	11.4%		
Energy use	MJ	2.08e+04	2.84e+03	4.68e+03	64.9%		
Solid waste landfill use	m3	1.67e-01	5.43e-02	5.47e-02	0.74%		
Hazardous waste landfill use	m3	1.68e-02	3.60e-03	3.60e-03	-0.01%		
Radioactive waste landfill use	m3	2.00e-04	1.00e-04	1.00e-04	-3.88%		
Global warming	kg-CO2 eq.s	6.95e+02	5.93e+02	6.17e+02	4.05%		
Ozone depletion ^a	kg-CFC-11 eq.s	2.05e-05	1.37e-05	1.37e-05	-0.26%		
Photochemical smog	kg-ethene eq.s	1.71e-01	1.41e-01	1.61e-01	13.7%		
Acidification	kg-SO2 eq.s	5.25e+00	2.96e+00	3.00e+00	1.48%		
Air particulates	kg	3.01e-01	1.15e-01	1.19e-01	3.85%		
Water eutrophication	kg-phosphate eq.s	4.82e-02	4.96e-02	4.96e-02	0.00%		
BOD	kg	1.95e-01	2.83e-02	2.83e-02	-0.12%		
TSS	kg	8.75e-01	6.15e-02	6.13e-02	-0.30%		
Radioactivity	Bq	3.85e+07	1.22e+07	1.22e+07	-0.09%		
Chronic health effects,	tox-kg	9.34e+02	6.96e+02	7.66e+02	10.1%		
occupational							
Chronic health effects, public	tox-kg	1.98e+03	9.02e+02	8.82e+02	-2.14%		
Aesthetics (odor)	m3	7.58e+06	5.04e+06	5.04e+06	0.01%		
Aquatic toxicity	tox-kg	2.25e-01	5.19e+00	5.19e+00	0.02%		
Terrestrial toxicity	tox-kg	1.97e+03	8.94e+02	8.74e+02	-2.25%		

 Table 3-61. Baseline and sensitivity analysis results—LCD modified module energy

^a LCD impacts in this category are greater than CRT impacts when phased out substances are removed from the inventories (see Section 3.3.6).

Figure 3-28 presents the LCD baseline and sensitivity analysis results for the energy use impact category, the category with the greatest percent change from the baseline to the modified LCD module energy scenario. Under this scenario, LCD energy use impacts in the manufacturing stage increased almost 230% from 1,440 MJ per functional unit to 3,280 MJ per functional unit. However, total life-cycle energy use impacts increased only 65%. This sensitivity analysis did not affect consumption rates outside of the manufacturing stage.



Figure 3-29 shows the effects of the sensitivity analysis on LCD chronic occupational health effects, another impact category with a relatively large percentage change. As shown in the figure, the manufacturing stage impact score in this category increased about ten percent, from 684 tox-kg per monitor to 755 tox-kg per monitor, due to the increase in fuel inputs. The chronic occupational health effect impacts were less sensitive than energy impacts because health effects results are calculated using a scoring approach that considers the inherent toxicity of a chemical instead of a simple loading approach (as is used for energy impacts). No toxicity data were available for LNG, the input with the greatest change in quantity. Therefore, the LNG HV is representative of a mean toxicity value.



3.4.4 Modified LCD EOL Dispositions Scenario

Finally, because very few desktop LCDs have reached their end of life, and usually only if they have been damaged in some way, very little is known about the EOL disposition of LCDs. In the baseline scenario it was assumed that a certain percent of EOL LCDs are incinerated, recycled, remanufactured, landfilled as solid waste, and landfilled as hazardous waste. (See Section 2.7.3 and Appendix I for an explanation of how EOL disposition percentages were determined.) To address uncertainties in the allocation of disposition numbers, as follows:

- change percent recycled from 15% to 0%,
- change percent remanufactured from 15% to 40%,
- change percent landfilled (solid waste) from 50% to 40%.
- do not change fraction incinerated (15%) or fraction sent to a hazardous waste landfill (5%).

Thus, under the modified EOL disposition scenario, recycling and solid waste landfilling impacts would decrease, remanufacturing impacts would increase, and incineration and hazardous waste landfilling impacts would not change. However, in attempts made to obtain remanufacturing data, it was found that remanufacturing processes spanned a wide range of activities, from as little as replacing button tops to as extensive as testing and replacing PWBs or transformers. Given the broad range of possibilities, and because few desktop LCDs have reached their end of life, no single set of operations could be identified to adequately represent remanufacturing activities that could be incorporated in our model. Remanufacturing data were, therefore, excluded from the assessment.

As shown in the baseline LCIA results (Section 3.3), LCD EOL dispositions have little effect on overall life-cycle impacts under the baseline scenario. In fact, the only impact categories in which an EOL process was a top contributor to overall impacts were the hazardous waste landfill use impact category, where the portion of a monitor landfilled contributed 97% of impacts, and the solid waste landfill use category, where the portion of a monitor landfilled contributed 3.5% of impacts. As noted above, hazardous waste landfill use impacts would not change under the modified LCD EOL dispositions scenario, but solid waste landfill impacts would be expected to decrease slightly. In a preliminary quantitative analysis of this scenario, LCD life-cycle solid waste landfill impacts were found to decrease less than one percent, and life-cycle impacts in other impact categories decreased less than 0.1%. Thus, the modified LCD EOL dispositions scenario would have only a minor effect on LCD life-cycle impacts and would not change comparative CRT and LCD results.

3.4.5 Summary of CRT and LCD Sensitivity Analysis Results

The results of the sensitivity analyses are useful to manufacturers who want to understand how uncertainty in the inventory affects impacts. This information can be used to identify areas for additional study or potential improvement opportunities. As discussed in Sections 3.4.1 through 3.4.2, it appears that CRT life-cycle impacts are highly sensitive to the glass energy data, and less sensitive to the lifespan assumptions (lifespan assumptions greatly affect the magnitude of CRT life-cycle impacts, but they do not greatly affect the distribution of impacts among life-cycle stage). LCD impacts are less sensitive to the glass energy data and in fact are not greatly affected by any of the sensitivity analysis scenarios, except the longer lifespan under the manufactured life scenario.

Sensitivity results are also useful to interested members of the public who may be evaluating the relative impacts of different monitor types and are interested in whether the CRT or LCD has greater life-cycle impacts in any given impact category. Table 3-62 presents the monitor type with greatest impacts by impact category and by scenario. This information helps us determine whether major assumptions (e.g., the monitor lifespan and LCD EOL distribution assumptions) or uncertain data (e.g., glass energy data and LCD monitor manufacturing energy) are driving results. As shown in the table, the modified glass energy scenario is the only scenario that significantly changes the results from the baseline CRT and LCD comparative results. Under this scenario, life-cycle impact results in seven categories reverse direction from the baseline assessment, such that the LCD has greater impacts than the CRT. Therefore, under this scenario, a total of nine out of 20 categories are greater for the LCD than the CRT, compared to two out of 20 categories under the baseline scenario. The only other scenario that affects these results is the manufactured life scenario, when impacts in the water eutrophication category are greater for the CRT than the LCD.

Impact category	Monitor type with greatest impacts by scenario							
	Baseline	Manu- factured life	Modified glass energy	Modified LCD module	Modifed LCD EOL distribution ^a			
				energy				
Renewable resource use	CRT	CRT	CRT	CRT	CRT			
Nonrenewable resource use	CRT	CRT	LCD	CRT	CRT			
Energy use	CRT	CRT	CRT	CRT	CRT			
SW landfill use	CRT	CRT	CRT	CRT	CRT			
HW landfill use	CRT	CRT	CRT	CRT	CRT			
RW landfill use	CRT	CRT	CRT	CRT	CRT			
Global warming	CRT	CRT	LCD	CRT	CRT			
Ozone depletion ^b		b b	b b					
Photochemical smog	CRT	CRT	LCD	CRT	CRT			
Acidification	CRT	CRT	CRT	CRT	CRT			
Air particulates	CRT	CRT	CRT	CRT	CRT			
Water eutrophication	LCD	CRT	LCD	LCD	LCD			
Water quality, BOD	CRT	CRT	LCD	CRT	CRT			
Water quality, TSS	CRT	CRT	LCD	CRT	CRT			
Radioactivity	CRT	CRT	CRT	CRT	CRT			
Chronic health effects, occupational	CRT	CRT	LCD	CRT	CRT			
Chronic health effects, public	CRT	CRT	CRT	CRT	CRT			
Aesthetics (odor)	CRT	CRT	LCD	CRT	CRT			

Table 3-62. Summary of CRT and LCD LCIA results

Impact category Monitor type with greatest impacts by scenario									
Impact category	Womfor type with greatest impacts by scenario								
	Baseline	Manu- factured life	modified glass energy	Modified LCD module energy	LCD EOL distribution ^a				
Aquatic toxicity	LCD	LCD	LCD	LCD	LCD				
Terrestrial toxicity	CRT	CRT	CRT	CRT	CRT				

Table 3-62. Summary of CRT and LCD LCIA results

^a Based on a qualitative evaluation, not quantitative results.

^b CRT impacts are greater than LCD impacts in this category when all data are included in the inventories, including data for substances that have been phased out. However, LCD impacts are greater than CRT impacts when phased out substances are removed from the inventories (see Section 3.3.6).

REFERENCES

- Bare, J.C., D.W. Penninton, and H.A. Udo de Haes. 1999. *Life Cycle Impact Assessment Sophistication: International Workshop. Int. J. LCA.* **4**(5): 299-306.
- Bare, J. 1999. Slide Presentation: "Consistent Science for Environmental Decision-making." Tools for Sustainability Workshop, February 18-19. Accessed at http://www.epa.gov/ORD/NRMRL/std/mtb/PDONE/TFS6.PDF. Accessed August 8, 2001.
- Barnthouse, L., J. Fava, K. Humphreys, R. Hunt, L. Laibson, S. Noesen, J. Owens, J. Todd, B. Vigon, K. Weitz, J. Young (Eds.). 1997. *Life-Cycle Impact Assessment: The State-of-the-Art*. Society of Environmental Toxicology and Chemistry. Pensacola, FL.
- Bintein, S., J. Devillers, and W. Karcher. 1993. "Non-linear Dependence of Fish Bioconcentration on n-Octonol/Water Partition Coefficient." SAR QSAR Environ. Res. 1:29-39.
- CAAA (Clean Air Act Amendments). 1990. Title VI (42 U.S. Code Annotated, Section 7671a; and associated regulations: 40CFR82, App. F, Subpart A and App. I, Subpart A).
- Curran, M.A. 1996. Environmental Life Cycle Assessment. McGraw-Hill: New York.
- EPA (Environmental Protection Agency). 1992. Reference Guide to Odor Thresholds for Hazardous Air Pollutants listed in the Clean Air Act Amendments of 1990. EPA/600/R-92/047. March.
- EPA. 1996. Compilation of Air Pollutant Emission Factors (AP-42), Fifth Edition, Volume I. January.
- EPA. 1998. National Air Quality and Emissions Trends Report, 1997. EPA/454/R-98-016. December.
- EPA. 1999. Integrated Risk Information System (IRIS). Database available at http://www.epa.gov/ngispgm3/iris/sybst-fl.htm.
- EPA. 2001a. EPA Web site (a), Methyl Bromide Phase Out information available at http://www.epa.gov/docs/ozone/mbr/mbrqa.html#q3. Accessed July 17, 2001.
- EPA. 2001b. EPA Web site (b), HCFC phaseout schedule available at http://www.epa.gov/ozone/title6/phaseout/hcfc.html. Accessed July 25, 2001.
- Fava, J., F. Consoli, R. Denison, K. Dickson, T. Mohin, and B. Vigon (Eds.). 1993. A Conceptual Framework for Life-Cycle Impact Assessment. Society of Environmental Toxicology and Chemistry (SETAC) and SETAC Foundation for Environmental Education. Pensacola, FL.

- Glazebrook, B. 2001. Electronic Communication with B. Glazebrook, Ecobilan and L. Kincaid, University of Tennessee, Center for Clean Products and Clean Technologies. July 9.
- Glazebrook, B. (b). 2001. Electronic Communication with B. Glazebrook, Ecobilan and L. Kincaid, University of Tennessee, Center for Clean Products and Clean Technologies. July 9.
- Guinee, J., R. Heijungs, L. van Oers, D. van de Meent, T. Vermeire, M. Rikken. 1996. *LCA Impact Assessment of Toxic Releases*. The Hague, The Netherlands.
- Harris, J.C. 1981. "Rate of Hydrolysis." In W. J. Lyman, W. F. Reehl and D. H. Rosenblatt (Eds.). Research and Development of Methods for Estimating Physicochemical Properties in Organic Compounds of Environmental Concern. Final Report, Phase II. Part I, Chapter 7, pp.1-48. Arthur D. Little, Cambridge, MA.
- Hauschild, M. and H. Wenzel. 1997. *Environmental Assessment of Products, Volume 2: Scientific Backgrounds*. London: Chapman and Hall.
- Haywood, B. 1999. University of Tennessee Technical Memorandum for Development of the Dermal Hazard Value.
- Heijungs, R., J.B. Guinee, G. Huppes, R.M. Lankreijer, H.A. Udo de Haes, A. Wegener Sleeswijk, A.M.M. Ansems, P.G. Eggels, R. van Duin, and H.P. De Goede. 1992. *Environmental Life-Cycle Assessment of Products. Vol. I: Guide*, and *Vol. II: Backgrounds*. Leiden: CML Center for Environmental Studies, Leiden University.
- Houghton, J. T. et al. (Eds.). 1996. Climate Change 1995: The Science of Climate Change. Contribution of Working Group to the Second Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, New York, NY.
- Hunter, R. S. and F. D. Culver. 1992. MICROQSAR version 2.0. Institute for Biological and Chemical Process Analysis, Montana State University, Bozeman, MT.
- IARC (International Agency for Research on Cancer). 1998. IARC Monographs. Database on Risks to Humans available at http://www.iarc.fr.
- IISI (International Iron and Steel Institute). 2001. World Steel Trends and Statistics. Web site available at http://www.worldsteel.org/trends_indicators/figures_3.html. Accessed July 31.
- International Life Sciences Institute (ILSI). 1996. *Human Health Impact Assessment in Life Cycle Assessment: Analysis by an Expert Panel*. Washington, DC.

- IPCC (Intergovernmental Panel on Climate Change). 1995. "IPCC Second Assessment Synthesis of Scientific-Technical Information Relevant to Interpreting Article 2 of the UN Framework Convention on Climate Change." Web site available at http://www.ipcc.ch/pub/sarsyn.htm. Accessed August 23, 2001.
- IUCLID (International Uniform Chemical Information Database). 1996. European Chemical Bureau, European Commission-JRC, Environment Institute, Ispra, Italy, 1996.
- Lindfors, L.G., K. Christiansen, L. Hoffman, Y. Virtanen, V. Juntilla, O-J Hanssen, A. Roenning, T. Ekvall. 1995. Nordic Guidelines on Life-Cycle Assessment. Copenhagen: Nordic Council of Ministers, Report # Nord 1995:20.
- Niemi, G.J., G.D. Veith, R.R. Regal, and D.D Vaishnav. 1987. "Structural Features Associated with Degradable and Persistent Chemicals." *Environ. Toxicol. Chem.* **6**:515-527.
- Penman, J., D. Kruger, I. Galbally, T. Hiraishi, B. Nyenzi, S. Emmanul, L. Buendia, R. Hoppaus, T. Martinsen, J. Meijer, K. Miwa, and K. Tanabe (Eds). 2000. Good Practice Guidance and Uncertainty Management in Natural Greenhouse Gas Inventories. IPCC National Greenhouse Gas Inventories Programme, Published for the IPCC by the Institute for Global Environmental Strategies, Japan.
- Smrchek, J. EPA. 1999. Electronic communication with J. Smrchek, EPA, and Dipti Singh, EPA. March 29.
- Swanson, M.B., G.A. Davis, L.E. Kincaid, T.W. Schultz, J.E. Bartmess, S.L. Jones, E.L. George. 1997. "A Screening Method for Ranking and Scoring Chemicals by Potential Human Health and Environmental Impacts." *Environmental Toxicology and Chemistry*. 16(2): 372-383.
- Swanson, M.B. and Adam C. Socha (Eds.). 1997. *Chemical Ranking and Scoring: Guidelines* for Relative Assessments of Chemicals. SETAC Press.
- Swanson, M.B., G.A. Davis, J.G. Overly, R. Dhingra, K. Kelly. 2001. Life Cycle Design for the Automobile: Environmental Technology Draft Report.

this page intentionally left blank

Chapter 4

QUALITATIVE RISK SCREENING OF SELECTED MATERIALS

The scope of the DfE CDP, as presented in Chapter 1, was to first conduct an environmental life-cycle assessment (LCA) to evaluate a generic 17" CRT and 15" LCD desktop computer display, followed by a streamlined Cleaner Technologies Substitutes Assessment (CTSA), which would target specific materials or processes that warrant further evaluation. Traditionally, the DfE Program has conducted CTSAs that perform detailed risk characterizations of alternative chemical processes. The streamlined CTSA for the CDP takes a more detailed look than the LCA at the toxic effects of chemicals used in a process, without conducting a complete risk characterization typical of past CTSAs.

In order to provide meaningful and timely results, the CDP Core group agreed to select a few materials that are of interest to EPA and industry and conduct analyses concurrently with the LCA, instead of conducting a CTSA on selected materials or processes after the LCA results were presented. The LCA identifies material inputs and outputs and then characterizes them in a life-cycle impact assessment (LCIA). In the human and environmental health effects impact categories, these input and output amounts are used as surrogates for exposure. For the selected materials, the additional CTSA-related analyses are intended to better understand the potential exposures to selected materials, during any processes that use thee materials, in order to try to better understand potential chemical risks.

The materials that were selected for further analysis were lead, mercury, and liquid crystals. The justification for choosing these materials, and a brief description of the scope are provided below:

- Lead: Lead is a top priority toxic material at the U.S. EPA. Lead is found in glass components of CRTs, as well as in electronics components (printed wiring boards and their components) of both CRTs and LCDs. The electronics industry is also concerned with lead and continues to take steps to reduce the use of lead in electronics products. Lead has been extensively studied for its toxic effects and has been addressed elsewhere with regard to CRTs and electronics (ATSDR, 1999). Therefore, within the scope of the CDP, lead is recognized as a material of concern, but extensive evaluation beyond the LCA is not conducted. Some discussion of the potential for exposure is included in this chapter, but it references existing studies for further information.
- **Mercury**: Another top priority toxic material at the U.S. EPA is mercury. The fluorescent tubes that provide the source of light in the LCD contain mercury. Although very small amounts of mercury are found in the LCD backlights, EPA's concern with mercury and the potential for exposure during manufacturing and end-of-life processes are reasons why a more detailed analysis of mercury is warranted in the CDP. In addition, mercury is emitted from some fuel combustion processes, such as coal-fired electricity generation processes, and there is interest in the relative magnitude of mercury emissions from these sources as compared to the magnitude of mercury is to begin to do an improvement assessment, because there appear to be potential alternatives to backlights with mercury.

4. QUALITATIVE RISK SCREENING OF SELECTED MATERIALS

• **Liquid crystals**: The toxicity of the liquid crystals in LCDs has been alluded to in the literature and there is a need to better understand the toxicity of these materials as well as provide the appropriate context of potential exposure and any associated risk. For example, during normal use of a display, no exposure would be expected. Liquid crystals are generally organic materials in broad categories such as polycyclic aromatic hydrocarbons (e.g., phenylcyclohexanes, biphenyls) (EIAJ, 1996). By including liquid crystals in a more detailed analysis, this chapter attempts to better characterize any potential hazard and/or potential exposure of liquid crystals from the manufacturing, use, and disposal of LCD monitors.

Choosing these three materials as a priority does not presume that these are the only materials of importance and worthy of additional analyses. The results of the LCI and LCIA in Chapters 2 and 3 provide more information on where to focus additional analysis efforts or improvements. Chapter 5 also identifies some potential improvement opportunities.

Sections 4.1 through 4.3 present the qualitative risk screening of lead, mercury, and liquid crystals, respectively. Subsections in each section briefly describe the following:

- *The use of the materials in computer displays:* These subsections describe how the materials are used in computer displays and give information on the mass used in particular applications.
- *Life-cycle inputs and outputs of the materials from computer displays:* The life-cycle assessment approach not only focuses on the material contained within the product, but also emphasizes the environmental impacts of material inputs and outputs from every life-cycle stage. These subsections summarize the life-cycle inputs and outputs of the materials found in the CDP life-cycle inventories (LCIs).
- *Life-cycle impacts associated with the material inputs and outputs:* As discussed in Chapter 3, life-cycle impact assessment (LCIA) is a screening-level evaluation of potential impacts to any system (e.g., the environment) as a result of some action (e.g., a chemical release). In the LCIA, life-cycle inventory data were classified into various impact categories (e.g., greenhouse gases or ozone depletion) based on the characteristics of the inventory item. Characterization methods were then used to quantify the magnitude of the contribution the emission or consumption of the inventory item could have in producing the associated impact. The result is expressed as an impact score which has been calculated using specific impact assessment tools. These subsections summarize the CDP LCIA results for lead, mercury, and liquid crystal inputs and outputs.
- *Potential exposures to the material including occupational, public, and ecological exposures:* Toxic materials may pose a threat to human health anytime there is the potential for human exposure throughout the life cycle of a computer display. Exposure occurs anytime a chemical or physical agent comes into contact with an organism, be it human or ecological. The magnitude of exposure depends on the concentration of the chemical at the contact point, and the duration and frequency of the exposure. The concentration of the chemical at the contact point is influenced by several factors, including the type, quantity, and disposition (e.g., airborne, surface water) of the initial release, and the subsequent environmental fate of the chemical (the ultimate disposition of the chemical as it is transported through the environment). Note that exposure is often defined in terms of exposure pathways. An exposure pathway describes the route a

contaminant travels from its source to an individual. A complete exposure pathway consists of the source and mechanism of release, transport medium, point of potential human contact, and the exposure route (e.g., inhalation, ingestion, dermal contact). These subsections focus on identifying potential exposures for three groups: workers in facilities using the chemicals (occupational exposures), the general public, who may be exposed to releases of the chemicals into the ambient environment, and ecological populations. While a quantitative exposure assessment is beyond the scope of this study, a qualitative discussion of potential pathways of exposure is presented for each of the groups listed above.

- *Potential human health effects:* Human health effects can include acute effects from short-term exposure, as well as chronic effects from repeated, long-term exposure. To be consistent with the scope of the LCIA, these subsections focus on chronic effects, including noncancer and cancer effects, unless no chronic toxicity data are available.
- U.S. environmental regulations for the material: These subsections briefly summarize U.S. environmental regulations that may affect facilities that manufacture materials for the computer display or are otherwise affected by the life cycle of computer displays (e.g., disposal facilities). They do not summarize environmental regulations from other countries where displays or display components are manufactured, which may differ significantly from U.S. regulations.
- Alternatives to reduce the use of the material in computer displays: These subsections identify alternatives that can substitute for a material, or, in some cases, source reduction methods to reduce the use of a material. The discussion of alternatives presented here is not a rigorous evaluation of their performance, cost, and environmental attributes, but rather a summary of the current knowledge base that may be useful for manufacturers seeking to identify improvement opportunities.

In Section 4.4, summary information and conclusions are presented.

4.1 LEAD

Lead and/or lead compounds are present in both CRTs and LCDs. Because lead and leadcontaining compounds have long been known to pose a threat to both human health and the environment, this section presents a more detailed look at the use of lead in computer displays and its effects on human health and the environment.

4.1.1 Lead in Computer Displays

Lead is a significant material in current CRTs, accounting for up to 8% of the overall composition of the CRT by weight (Menad, 1999), with a 17" monitor containing as much as 1.12 kg of lead (Monchamp *et. al.*, 2001). Lead is used in several parts of the CRT monitor, including the funnel and neck glass, the sealing frit, as solder on printed wiring boards (PWBs) within the monitor, and sometimes in the front panel glass of the CRT. Lead is not as prevalent in LCDs, only being found on PWBs.

Lead, in the form of lead oxide, lines the inner surface of both the neck and funnel glass of the CRT, or may in some cases be contained within the glass itself. The lead oxide layer acts as a shield, protecting users from x-ray emissions given off by the electron gun. The lead oxide layer can comprise as much as 28% by weight of the funnel (Lee *et al.*, 2000) and 32% of the neck (Menad, 1999).

The sealing frit, which is used to make a vacuum-tight connection between the funnel and the front panel, is comprised of as much as 80% lead oxide (Busio and Steigelmann, 2000). The lead oxide is mixed with boric oxide and zinc oxide, along with several other compounds, into a paste. The glass frit paste is applied as a bead around the top of edge of the funnel and allowed to dry. The front panel is then attached to the funnel and fired using a belt furnace for 30 to 60 minutes at typical temperatures of 450°C (Busio and Steigelmann, 2000). The lead, boric, and zinc oxides devitrify to form large crystals which give strength to the frit seal (Techneglas, 2001). Recent research has been directed at either reducing the lead content of the frit or reducing the energy required to fuse the frit.

Printed wiring boards found in both LCDs and CRTs primarily use a lead-based solder as a surface finish and to attach electrical components to the circuit board. Solder is typically comprised of 37-40% lead. Depending on the type of component, parts can be applied using a solder paste which is subsequently melted, or by passing the boards over a wave of molten solder. Data collected for the life-cycle inventory indicates that a 17" monitor has approximately 51 grams of solder, or just under 19.8 grams of lead, while a 15" LCD panel contains nearly 22.4 grams of solder, or just about 8.5 grams of lead. Several lead-free solder alternatives are currently being tested and are in limited use. Solder is the only significant source of lead in LCDs.

The CRT panel glass itself may also contain a small percentage of lead oxide, typically ranging up to 4% by weight (Lee *et al.*, 2000). The lead acts as a stabilizer for the glass during its formation, and also serves to keep the glass from browning. Lead-free CRT panel glasses are currently being produced successfully.

A list of the lead-containing parts that make up a computer display, along with the quantity of lead and percentage of lead in each part, is presented in Table 4-1.

Part	Display type	Quantity (Kg) ^a	% Lead content of part (by weight)						
Funnel	CRT	0.91	22-28% ^{b, c}						
Front panel	CRT	0.18	0-4 ^{b, c}						
Neck	CRT	0.012	26-32 ^{b, c}						
Frit	CRT	0.026	70-80 ^{b, c, d}						
PWBs (total)	CRT	0.051	N/A						
PWBs (total)	LCD	0.043	N/A						

Table 4-1. Computer display parts that contain lead

^a Quantity of lead in a 17" monitor (Monchamp et. al., 2001).

^b Menad, 1999.

^c Lee *et. al.*, 2000.

^dBusio and Steigelmann, 2000.

N/A = Not applicable

4.1.2 Life-Cycle Inventory Inputs and Outputs of Lead for Computer Displays

Data on lead and lead-containing materials were collected and compiled as part of the life-cycle inventory. Material inputs containing lead included primary materials (e.g., lead-based solder) which end up as part of the product, as well as from ancillary materials (e.g., lead consumed during the production of steel used to make CRT parts) which are consumed as part of the manufacturing process or other supporting processes, such as energy production. The data were aggregated by material from individual processes and are presented by life-cycle stage for both CRTs and LCDs in Tables 4-2 and 4-3 below. More detailed material input data, which include the processes for each input and the quantity of lead released, are included in Appendix N.

Tuste : 20 offer four containing inputs by ine cycle stage								
Life-cycle stage	Input	Quantity	Units	Туре				
Materials processing	Lead (Pb, ore)	6.50E-05	kg	Ancillary material				
Materials processing	Lead (Pb, ore)	4.96E-01	kg	Primary material				
Manufacturing	Frit	6.67E-02	kg	Primary material				
Manufacturing	Lead	4.94E-01	kg	Primary material				
Manufacturing	Printed wiring board (PWB)	8.47E-01	kg	Primary material				
Manufacturing	Solder (63% tin; 37% lead)	5.08E-02	kg	Primary material				
Manufacturing	Solder, unspecified - (CRT Assembly)	2.67E-02	kg	Primary material				

 Table 4-2. CRT lead-containing inputs by life-cycle stage

Life-cycle stage	Input	Quantity	Units	Туре
Materials processing	Lead (Pb, ore)	2.47E-05	kg	Ancillary material
Manufacturing	Printed wiring board (PWB)	3.74E-01	kg	Primary material
Manufacturing	Solder (60% tin, 40% lead)	3.81E-02	kg	Primary material
Manufacturing	Solder (63% tin; 37% lead)	2.24E-02	kg	Primary material
Manufacturing	Solder, unspecified	7.35E-05	kg	Ancillary material

 Table 4-3. LCD lead-containing inputs by life-cycle stage

Material inputs can be raw materials such as lead ore, or output materials from a previous process or life-cycle stage. For example, small quantities of lead extracted from lead ore are sometimes used as additives in the production of several materials including ferrite, steel, and invar. Once extracted, lead is an input material to the manufacturing processes of several CRT components including CRT glass manufacturing and the manufacturing of the sealing frit paste, which then is an input for the CRT tube manufacturing process. Similarly, lead is used to produce solder which is then used to produce PWBs used in LCDs and CRTs.

Releases of lead and lead-based materials into the environment occur throughout the entire life cycle of the computer display. Environmental releases include airborne, waterborne, solid waste, and radioactive emissions of lead isotopes associated with nuclear fuel reprocessing. Similar to the inputs, emissions data were aggregated by the material released from individual processes and then reported by life-cycle stage. The lead or lead-based material released, the quantity of the release, the type of release (e.g., waterborne), and the ultimate disposition of the release all contribute to the environmental impacts.

The life-cycle outputs containing lead for both CRTs and LCDs are shown in Tables 4-4 and 4-5, respectively. More detailed data on lead and lead-based outputs for each process are presented in Appendix N.

Life-cycle stage	Output	Quantity	Units	Туре	Disposition
Materials processing	Lead	1.66E-03	kg	Airborne	Air
Materials processing	Lead	2.29E-08	kg	Solid waste	Landfill
Materials processing	Lead compounds	1.59E-05	kg	Waterborne	Surface water
Materials processing	Lead-210 (isotope)	1.02E+00	Bq	Radioactivity	Air
Manufacturing	Broken CRT glass	1.88E-03	kg	Hazardous waste	Landfill
Manufacturing	Broken CRT glass	1.08E+00	kg	Solid waste	R/R
Manufacturing	Cinders from CRT glass mfg (70% PbO)	8.26E-03	kg	Hazardous waste	Landfill
Manufacturing	CRT glass faceplate EP dust (Pb) (D008 waste)	1.03E-03	kg	Hazardous waste	Landfill
Manufacturing	CRT glass funnel EP dust (Pb) (D008 waste)	5.01E-03	kg	Hazardous waste	R/R
Manufacturing	Frit	2.99E-03	kg	Hazardous waste	Landfill
Manufacturing	Hazardous sludge (Pb) (D008)	1.52E-03	kg	Hazardous waste	Landfill
Manufacturing	Lead	1.03E-06	kg	Waterborne	Treatment
Manufacturing	Lead	1.30E-05	kg	Airborne	Air
Manufacturing	Lead	4.64E-05	kg	Waterborne	Surface water
Manufacturing	Lead (Pb, ore)	4.41E-07	kg	Airborne	Air
Manufacturing	Lead compounds	1.62E-05	kg	Waterborne	Treatment
Manufacturing	Lead compounds	1.17E-09	kg	Waterborne	Surface water
Manufacturing	Lead contaminated grit (D008 waste)	3.46E-05	kg	Hazardous waste	Landfill
Manufacturing	Lead debris (D008 waste)	2.14E-04	kg	Hazardous waste	Landfill
Manufacturing	Lead sulfate cake	2.67E-05	kg	Hazardous waste	Landfill
Manufacturing	Printed wiring board (PWB)	3.70E-02	kg	Solid waste	R/R
Manufacturing	PWB-Solder dross	6.70E-02	kg	Hazardous waste	R/R
Manufacturing	Sludge from CRT glass mfg (1% PbO)	8.77E-04	kg	Hazardous waste	Landfill
Manufacturing	Waste batch (Ba, Pb) (D008 waste)	1.41E-03	kg	Hazardous waste	Landfill
Manufacturing	Waste finishing sludge (Pb) (D008 waste)	2.56E-04	kg	Hazardous waste	Landfill
Use	Lead	1.27E-05	kg	Airborne	Air
End-of-life	Lead	1.42E-05	kg	Airborne	Air
End-of-life	Lead compounds	1.60E-09	kg	Waterborne	Surface water
End-of-life	Printed wiring board (PWB)	1.46E-01	kg	Hazardous waste	R/R

 Table 4-4. Life-cycle lead outputs to the environment from CRTs

R/R = recycling/reuse

Life-cycle stage	Outputs	Quantity	Units	Туре	Disposition
Materials processing	Lead	3.13E-06	kg	Airborne	Air
Materials processing	Lead	5.42E-09	kg	Solid waste	Landfill
Materials processing	Lead compounds	3.68E-06	kg	Waterborne	Surface water
Materials processing	Lead-210 (isotope)	3.21E-01	Bq	Radioactivity	Air
Manufacturing	Lead	8.84E-06	kg	Airborne	Air
Manufacturing	Lead	8.33E-07	kg	Waterborne	Treatment
Manufacturing	Lead (Pb, ore)	1.48E-06	kg	Airborne	Air
Manufacturing	Lead compounds	5.67E-11	kg	Waterborne	Surface water
Manufacturing	Lead compounds	7.14E-06	kg	Waterborne	Treatment
Manufacturing	Printed wiring board (PWB)	7.50E-03	kg	Solid waste	Landfill
Manufacturing	PWB-Lead contaminated waste oil	5.14E-03	kg	Hazardous waste	Treatment
Manufacturing	PWB-Solder dross	2.96E-02	kg	Hazardous waste	Recycling/
_			_		reuse
Manufacturing	Waste batch (Ba, Pb) (D008 waste)	6.55E-05	kg	Hazardous waste	Landfill
Manufacturing	Waste CCFL, with lead	8.17E-08	kg	Hazardous waste	Treatment
Use	Lead	4.76E-06	kg	Airborne	Air
End-of-life	Lead	4.76E-06	kg	Airborne	Air
End-of-life	Lead compounds	4.98E-10	kg	Waterborne	Surface water

 Table 4-5. Life-cycle lead outputs to the environment from LCDs

4.1.3 Computer Display Life-Cycle Impacts for Lead

The life-cycle impacts of lead, lead compounds, and materials containing lead (e.g., leadbased solder on printed wiring boards) calculated for CRTs and LCDs during the LCIA are summarized in Tables 4-6 and 4-7 respectively. Impact scores in the table are expressed in units specific to each impact category (see Chapter 3.1 for a discussion of impact category units and weighting). The total impact score for each category resulting from lead and lead-based materials is presented at the bottom of each table.

			Impact Scores by Category							
Life-cycle stage	Material	Non- renewable resource (kg)	Hazardous waste landfill use (m ³)	Solid waste landfill use (m ³)	Radio- activity (Bq)	Chronic health effects- public	Chronic health effects- occupational	Aquatic toxicity (tox-kg)	Terrestrial toxicity (tox-kg)	
						(tox-kg)	(tox-kg)			
	Lead	0	0	0	0	3.31e-03	0	0	1.66e-03	
Materials	Lead (Pb, ore)	4.96e-01	0	0	0	0	0	0	0	
processing	Lead compounds	0	0	0	0	3.17E-05	0	3.10e-04	1.59E-05	
	Lead-210 (isotope)	0	0	0	1.02E+00	0	0	0	0	
	Broken CRT glass	0	6.22E-07	0	0	0	0	0	0	
	Cinders from CRT glass mfg (70% PbO)	0	6.88E-06	0	0	0	0	0	0	
	CRT glass faceplate EP dust (Pb) (D008 waste)	0	2.15E-06	0	0	0	0	0	0	
	Frit	0	3.04E-06	0	0	0	0	0	0	
	Hazardous sludge (Pb) (D008)	0	1.38E-06	0	0	0	0	0	0	
Manufacturing	Lead	4.94e-01	0	0	0	1.19e-04	9.88e-01	9.3E-05	5.94E-05	
Wanutacturing	Lead compounds	0	0	0	0	2.35E-09	0	2.3E-08	1.17E-09	
	Lead contaminated grit (D008 waste)	0	2.99E-09	0	0	0	0	0	0	
	Lead debris (D008 waste)	0	1.85E-08	0	0	0	0	0	0	
	Lead sulfate cake	0	3.03E-08	0	0	0	0	0	0	
	Sludge from CRT glass mfg (1% PbO)	0	6.45E-07	0	0	0	0	0	0	
	Waste Batch (Ba, Pb) (D008 waste)	0	1.22E-07	0	0	0	0	0	0	
	Waste finishing sludge (Pb) (D008 waste)	0	2.32E-07	0	0	0	0	0	0	
Use	Lead	0	0	0	0	2.55E-05	0	0	1.27E-05	
End-of-life	Lead	0	0	0	0	2.85E-05	0	0	1.42E-05	
	Lead compounds	0	0	0	0	3.19E-09	0	3.1E-08	1.6E-09	
Total Impact S	cores By Category	9.89e-01	1.52E-05	0	1.02E+00	3.52e-03	9.88e-01	4.00e-04	1.80e-03	

Table 4-6. Summary of Lead Impact Scores for CRTs

Life-cycle stage	Material	Impact scores by category							
		Non- renewable resource (kg)	Hazardous waste landfill use (m ³)	Solid waste landfill use (m ³)	Radio- activity (Bq)	Chronic health effects- public	Chronic health effects- occupational	Aquatic toxicity (tox-kg)	Terrestrial toxicity (tox-kg)
						(tox-kg)	(tox-kg)		
Materials processing	Lead	0	0	0	0	6.26E-06	0	0	3.13E-06
	Lead (Pb, ore)	2.47E-05	0	0	0	0	0	0	0
	Lead compounds	0	0	0	0	7.36E-06	0	7.25E-05	3.68E-06
	Lead-210 (isotope)	0	0	0	3.21E-01	0	0	0	0
Manufacturing	Lead	0	0	0	0	1.77E-05	0	1.64E-05	8.84E-06
	Lead compounds	0	0	0	0	1.13E-10	0	1.12E-09	5.67E-11
	Printed wiring board (PWB)	0	0	9.38E-06	0	0	0	0	0
	Waste Batch (Ba, Pb) (D008 waste)	0	5.67E-09	0	0	0	0	0	0
Use	Lead	0	0	0	0	9.52E-06	0	0	4.76E-06
End-of Life	Lead	0	0	0	0	9.52E-06	0	0	4.76E-06
	Lead compounds	0	0	0	0	9.95E-10	0	9.80E-09	4.98E-10
Total Impact Scores By Category		2.47E-05	5.67E-09	9.38E-06	3.21e-01	5.03E-05	0	8.9E-05	2.52E-05

Table 4-7. Summary of lead impact scores for LCDs

Impact scores for some lead-based inputs and outputs shown in Tables 4-2 though 4-5 were not calculated if the type and disposition of the input or release was not expected to contribute to any of the impact categories. For example, a waterborne release of lead with a disposition going to treatment assumed that lead was not yet released to the environment where impacts could occur, and therefore no impacts were calculated. However, since inventory data for subsequent disposal processes could not be obtained, it was assumed the lead (or other inventory item) had been removed to a level such that the subsequent release of treated wastewater would not contribute significantly to the impact. Similarly, impact scores were not calculated for releases going to recycling/reuse or for product outputs.

Lead-based impacts from the CRT ranged from moderately to significantly greater than those from the LCD in every category, with the exception of solid waste landfill use. The most significant difference was in non-renewable resource consumption, where the CRT (989 grams) consumed over 40 thousand times the mass of non-renewable resources over the course of its life cycle than those consumed by the LCD (0.025 grams). Hazardous waste landfill use is another significant difference, with lead-based life-cycle outputs from CRTs using over 2,600 times the space of the lead-based outputs from LCDs. However, the absolute volume of waste from the CRT is still a relatively small volume (1.50 cm³). Other categories where CRTs had notably greater impacts as a result of lead include the chronic public health effects and terrestrial toxicity impact categories.

Based on the CDP LCIA methodology, chronic occupational health effect impacts were only calculated for lead inputs (excluding lead ore) to processes in the computer display life cycle. Only the manufacturing life-cycle stage had lead inputs from which impacts were calculated as shown in Table 4-6. The overall impact scores (0.988 tox-kg for CRT, none for LCD) likely underestimate the chronic occupational impacts for lead because they do not consider chronic occupational impacts from other processes such as the mining, smelting, and refining of the lead, which are known to pose potential occupational exposures (see Section 4.1.4). For a more detailed discussion of how chronic occupational health effect impacts were calculated, refer to Section 3.1.2.12.

The contribution of lead-based impacts for each computer display technology to the overall impacts for each individual impact category is shown in Table 4-8. Values in the table are expressed in the percent contribution the material made to the overall impact score for all materials (e.g., mercury, fuel oil, glass) for each category. The percent contributions give an indication of the importance of lead-based impacts relative to the life-cycle impacts from other materials or outputs from the computer display.

Impact category	CRT	LCD
Non-renewable resource	1.48E-01 %	6.78E-06 %
Hazardous waste landfill use	8.99E-02 %	1.57E-04 %
Solid waste landfill use	NA	1.73E-02 %
Radioactivity	2.70E-08 %	2.63E-06 %
Chronic health effects- public	1.80E-04 %	5.58E-06 %
Chronic health effects- occupational	1.10E-01 %	N/A
Aquatic toxicity	1.96E-01 %	1.71E-03 %
Terrestrial toxicity	8.90E-05 %	2.82E-06 %

 Table 4-8. Summary of percent contributions from lead-based materials to individual impact categories

N/A= Not applicable

It can be seen from Table 4-8 that the contributions of lead-based impacts are not significant relative to the total impacts from other materials (e.g., glass, copper wire, electronic components) in each category. Impacts from lead-based CRT outputs in the categories of nonrenewable resources, aquatic toxicity, and chronic public health effects are all range from 0.1-0.2% of the overall impact scores in each category.

4.1.4 Exposure Summary

Lead may pose a threat to human health anytime there is the potential for human exposure to the lead throughout the life cycle of a computer display. Exposure occurs anytime a chemical or physical agent, in this case lead or lead compounds, comes into contact with an organism, be it human or ecological. This section qualitatively identifies potential exposures for three groups: occupational workers in facilities using lead (occupational exposures), the general population living nearby these facilities which may be exposed to lead releases into the ambient environment, and ecological populations in the area surrounding a facility.

4.1.4.1 Occupational exposures

Workers are typically exposed to far greater concentrations of chemicals for longer periods of time than other populations. Worker exposures to lead can be especially serious given the overall toxicity of lead and lead compounds. As a result, both employers and government agencies have adopted recommendations or requirements for employers who wish to limit worker exposures.

Occupational exposures can occur anytime a worker comes into contact with lead, whether it be through dermal (skin) contact with a part containing lead (e.g., lead oxide coating on glass funnel), through the inhalation of lead particulates dispersed into the air, or through the inadvertent ingestion of lead. Many of the primary and support processes required to manufacture computer displays have lead in the workplace, and correspondingly, the potential for worker exposure. The processes associated with lead inputs and outputs throughout the computer displays' life cycles are presented in Tables N-1 through N-4 in Appendix N. It is important to note that while this list gives an indication of where likely lead exposures may occur, it is not exhaustive. Many processes and subprocesses may be contained within a process listed, each of which may pose its own potential for occupational lead exposures. Exposures to lead are more likely to occur during the extraction, manufacturing, and disposal life-cycle stages of a computer display. During the use of the computer display, potential exposures to lead are unlikely as the components containing the lead are contained within the outside shell of the computer display, limiting the opportunity for contact with consumers. Table 4-9 presents some typical pathways leading to the occupational exposure of workers to lead over the life cycle of a computer display.

Exposure route	Transport media	Example mechanisms of exposure
Inhalation	Air	Lead fumes resulting from the vaporization of lead during smelting
	Air	Lead oxide dust released to the air during lead frit manufacturing
	Air	Lead aerosols created during the aeration of tin/lead solder plating baths during PWB production
Dermal	Direct contact	Handling of leaded CRT glass funnels prior to assembly
Ingestion	Direct contact	Consumption of food eaten with lead-contaminated hands (or drinking, smoking, etc)
	Air	Ingestion of lead contaminated soil particles which become airborne during lead mining

Table 4-9.	Potential occupational exposure pathways for lead over the
	life cycle of a computer display

Workers may be exposed to airborne lead concentrations through the release of lead dust, fumes, or aerosols into the workplace. The lead is transported by the air, where it is inhaled into the lungs and then absorbed into the bloodstream. The greatest potential for high-level occupational exposure is during lead smelting and refining, where lead is vaporized during high temperature heating resulting in the release of lead fumes and small respirable particles of lead (EPA, 1986). Lead concentrations in air at three primary lead smelters were found to range from 80-2,900 μ g/m³, peaking at a level 58 times the OSHA recommended guidance level of 50 μ g/m³ (HSDB, 2001). Another study found that during the smelting and refining of lead, mean concentrations of lead in air reached as high as 4,470 μ g/m³, nearly 90 times the OSHA guidance level (Fu and Bofetta, 1995). Exposures to lead dust may also occur during lead mining, frit manufacturing, CRT glass manufacturing, or processes in which metallic lead is heated in the presence of air. Exposures to lead fumes are only possible during high temperature operations (above 500°C), such as welding or spray coating of metals with molten lead (Sittig, 1985).

Dermal exposures can take place anytime lead or materials containing lead are physically handled by workers. Opportunities for dermal exposures to lead are numerous in processes throughout the computer display life-cycle, as many processes involve lead or parts containing lead, especially in CRT manufacturing. Lead can be transferred to the skin of workers through contact with lead-containing materials and parts. Dermal exposures may also occur during cleaning and maintenance of equipment used to smelt, refine, or apply lead in a molten state (e.g., solder wave machinery for PWBs) or in areas with large airborne lead concentrations that may settle out onto work surfaces directly contacted by workers. The contribution of dermal exposures to the overall lead body burden is uncertain. It is believed that most forms of lead are unable to readily penetrate the skin, allowing only a small amount of lead to enter the bloodstream. (ATSDR, 1999). Alkyl lead compounds, which are the known exception, are primarily used as additives in gasoline and are not used directly in computer display manufacturing (Bress and Bidanset, 1991; ATSDR, 1999). Therefore, dermal exposures to inorganic lead compounds are not expected to be as significant as the inhalation or ingestion routes of exposure (EPA, 1986).

Along with inhalation, ingestion of lead-bearing dust and fumes is a major route of exposure in lead smelting and refining industries (EPA, 1986). Airborne dust particles of lead can eventually settle onto skin, equipment, clothing, and work surfaces, where they may be subsequently transferred to the mouth and become ingested. Airborne particles may also be inhaled and swallowed, directly when greater than 5 micrometers in size (ATSDR, 1999). Once ingested, the amount of lead that reaches the bloodstream through the stomach depends on a number of factors, such as the age of the subject, length of time since last meal, and how well the lead was able to dissolve in the stomach. Studies have found that roughly 6% of the lead ingested will absorb into the blood stream of an adult who has recently eaten (within the last day), while upwards of 60-80% was absorbed in adults who had not recently eaten (ATSDR, 1999).

Lead exposures of workers are frequently measured by biological testing (e.g., blood lead levels, urinary lead levels) rather than monitoring the workplace for lead concentrations, making occupational data on lead exposures often not readily available (EPA, 1986). For a discussion of blood lead levels, corresponding effects, and recommended exposure guidelines, refer to Section 4.1.5 of this chapter.

Blood-lead levels have been reported in studies of workers for several industries relevant to computer display manufacturing. For example, workers occupationally exposed to lead during glass production were tested to determine their blood lead levels. Workers were divided into groups based on work activities and blood samples were collected at the end of each shift. Concentrations of lead in the blood ranged from 70 to 680 μ g/l, with median values ranging from 170 to 340 μ g/m³, depending on the worker group. Data on types and rates of exposure were not identified (Ludersdorf *et al.*, 1987). Another study found that workers producing ceramic coated capacitors and resistors using leaded glass were exposed to occupational lead levels ranging from 61 to 1,700 μ g/m³. Blood lead levels ranged from 16 to 135 μ g/dL in these same workers, greatly exceeding the OSHA recommended level of 50 μ g/m³ (Kaye *et al.*, 1987).

The presence of lead in the workplace does not mean that occupational exposures are unavoidable. Worker exposures to lead can be reduced or even eliminated through the use of personal protective equipment, sound operating practices, or through advanced machinery that protects workers from exposure (e.g., an enclosed and vented wave solder machine). To determine actual worker exposures to lead, a complete exposure assessment specific to each manufacturing process would be required.

4.1.4.2 General population

The general population living nearby a manufacturing facility using lead may potentially be exposed to lead emissions from the facility into the surrounding ambient environment. The likelihood and quantity of the potential exposure is dependent on the type and quantity of release, the receiving media, the local environmental conditions, and the fate and transport characteristics of the release. General population exposure to lead is most likely to occur through ingestion of lead contaminated food, water, and soil, as well as through inhalation of lead particulates in the ambient air (EPA, 1986).

Lead released into the ambient air will typically be in the form of lead particulate matter, which is eventually removed from ambient air through washout by precipitation (rain or snow) or through gravitational settling. Estimates indicate that the majority of lead released into the environment is dispersed into the atmosphere (EPA, 1980). With a relatively small mass mean diameter of 0.55 μ m (HSDB, 2001), lead-containing particles can stay aloft for up to 64 hours and travel 1600 km, though they are more likely to be deposited within 10 km of the emission source (HSDB, 2001). General populations living near a source of lead emissions may encounter the lead while it is still airborne, leading to potential inhalation exposure. The direct inhalation of lead accounts for only a small part of the overall lead exposure to nearby populations, although the reentrainment of lead-contaminated soil is a common route of exposure (ATSDR, 1999).

Ingestion of lead is the most significant route of exposure for general populations (ATSDR, 1999). Particulates removed from the air are deposited into the soil, surface water, and onto local vegetation, where they may be ingested by nearby residents. Grains, vegetables, and fruits grown in close proximity to a source of lead emissions may contain lead which has been absorbed from contaminated soil through the root system. Lead also has the ability to bioaccumulate in the soft tissues of fish and wildlife, which are then consumed by sportsmen and their families.

Incidental ingestion of soil, which may occur while eating or smoking with soil-coated hands or when soil becomes reentrained and swallowed directly, often results in the largest lead exposures to residents living near emission sources. Lead-contaminated soil can also enter the home by being tracked into the house or carried home from the workplace on clothing, where it can come into contact with eating surfaces or food and become ingested. A study measuring lead in the home found mean lead levels as high as 22,191 μ g/g in homes located within 1.6 km of a lead smelting facility, and mean levels of 2,687 μ g/g in homes of workers at the smelting facility, irrespective of distance from the plant (ATSDR, 1999). One study found that once lead is swallowed, up to 50% of the lead is released from the contaminated soil into the stomach after only 10 minutes (HSDB, 2001).

Lead may also be released directly to surface water or indirectly to groundwater through the leaching of lead from landfills. Life-cycle inventory releases to surface water include 464 grams of lead and 159 grams of lead compounds per CRT. Surface water may also become contaminated through soil deposition or through surface water run-off from contaminated soil. Groundwater lead contamination from the leaching of lead-contaminated debris from solid- or hazardous waste disposal sites is unlikely to be significant due to the relative insolubility of lead (HSDB, 2001). Lead released to both surface waters and groundwater will typically remain insoluble, forming precipitates and settling into the sediment of the lake or stream. However, very little lead is typically found in U.S. waters which are used to supply the public with drinking water, due to strict governmental regulations (0.005 ppm lead) (ATSDR, 1999).

4.1.4.3 Ecological populations

Inorganic lead typically does not pose a significant health threat to fish and wildlife populations, except at extremely high concentrations. Once introduced into surface waters, the levels of soluble lead depend on the pH of the water and the dissolved salt content. At neutral pH, inorganic lead typically does not remain soluble in water at high concentrations, forming a precipitate that ultimately deposits in the sediment. However, as the alkalinity and pH decrease, the relative soluble concentrations of lead may become higher.

Toxic substances such as lead are capable of concentrating in the tissues of fish and wildlife. The bioconcentration of lead in fish is low-to-moderate in most species, with a bioconcentration factor (BCF) of 42 an 45 being reported for two fresh water fish species¹. However, BCFs for certain other species, such as blue mussles (4,985), eastern oysters (1,000+) and 4 types of fresh water invertebrate species (range of 499 to 1,700), were much higher (EPA, 1999).

4.1.5 Human Health Effects

Lead has been classified by EPA as a persistent biaccumulative toxic (PBT) chemical (EPA, 2001b). PBT pollutants are highly toxic, long-lasting substances that can build up in the food chain to levels that are harmful to human and ecosystem health. Lead's ability to persist in the environment without breaking down, along with its tendency to bioaccumulate, poses adverse health effects to birds and mammals at the top of the food chain, along with anyone who consumes them for food. Lead and lead-based compounds have been associated with a range of adverse human health effects, including effects on the nervous system, reproductive and developmental problems, and cancer.

4.1.5.1 Chronic effects (noncancer)

Lead is toxic to human health regardless of the form (Gosselin, 1984). It is one of the most hazardous of the toxic compounds because the dose of lead is cumulative over a lifetime, and the health effects are many and severe. Lead has been known to cause hematological, gastrointestinal, and neurological dysfunction in adults and children. Chronic exposures have also caused hypertension and reproductive impairment in both men and women, as well as slowed development in children (Sittig, 1985).

Adverse effects, other than cancer or mutations, are generally assumed to have a dose or exposure threshold. A reference dose (RfD) is an estimate of the daily exposure through ingestion to the human population that is likely to be without an appreciable risk of noncancer detrimental effects during a lifetime. Likewise, a reference concentration (RfC) represents an estimate of the daily inhalation exposure to the human population that is likely to be without an appreciable risk of noncancer appreciable risk of noncancer detrimental effects during a lifetime.

Because of the relative toxicity of lead and the cumulative nature of lead doses, a safe level of human exposure has yet to be identified by researchers, preventing EPA from

¹ Bioconcentration is defined by EPA as the non-dietary accumulation of chemicals in aquatic organisms (U.S. EPA, 1999).

establishing a RfD or RfC for inorganic lead (ATSDR, 1999). Instead, lead exposure is determined by using exposure biokinetic models that relate exposure levels to an estimated blood lead level, which is then compared to actual blood lead levels where adverse effects are known to occur². For example, increased blood pressure has been observed in adults with a blood-lead level as low as 7 μ g/dL (ACGIH, 1991). Lead concentrations in excess of 60 μ g/100g blood have been associated with neuropathy, gastrointestinal disturbances, and anemia, while workers with blood-lead levels between 50-70 μ g/100 g to have shown decreased neural response (ACGIH, 1991).

As a guideline, a blood-lead level of concern for adult workers of 30 μ g/dL has been established by both the Occupational Safety and Health Administration (OSHA) and ACGIH. A guideline of 10 μ g/m³ (for a child) has been set by the Center for Disease Control (CDC) for general population exposures to lead in the ambient environment. A summary of human health effect guidelines for lead is presented in Table 4-10.

Туре	Agency/Category	Regulatory level	
Workplace exposures to	lead		
Worker blood-lead	OSHA, Adults who "wish to bear children"	30 µg/dL	
target/action levels	OSHA, Blood-lead level of concern	40 µg/dL	
	OSHA, Medical removal	50 µg/dL	
	ACGIH, Biological Exposure Index (BEI) Blood-lead level of concern (ACGIH, 1998)	30 µg/dL	
	NIOSH, level to be maintained through air concentrations	60 µg/100 g	
Pregnant worker:	OSHA	30 µg/100 g	
target/action levels	CDC	10 µg/dLª	
Workplace air	OSHA Permissible exposure limit (PEL)	50 µg/m ³	
exposure limit	NIOSH Recommended exposure limit (REL) (NIOSH, 1997)	100 µg/m ³	
	ACGIH TLV TWA (ACGIH, 1998)	50 µg/m ³	
Ambient environment ex	xposures to lead		
Blood-lead	CDC	10 µg/m ^{3 a}	
target/action levels for	OSHA	30 µg/100 g	
child	World Health Organization blood-lead level of concern	$20 \mu g/dL$	

 Table 4-10. Human health effect regulations and guidelines for lead

^a CDC considers children to have an elevated level of lead if the amount of lead in the blood is at least 10 μ g/dL. Medical evaluation and environmental remediation should be done for all children with blood-lead levels greater than 20 μ g/dL. Medical treatment may be necessary for children with a blood-lead concentration above 45 μ g/dL (RTI, 1999).

Notes: ACGIH: American Conference of Governmental Industrial Hygienists; NIOSH: National Institute for Occupational Safety and Health; TWA: Time weighted average; TLV: Threshold limit value

² In order to estimate blood-lead levels, worker exposure levels based on releases reported in the inventory would be required. However, without information pertaining to the exposure conditions (which is unavailable to this study) and fate and transport of the releases, worker exposure cannot be calculated.

4.1.5.2 Carcinogenicity

The potential for a chemical to cause cancer is evaluated by weight-of-evidence classifications, specific to the rating organization, which are typically determined by laboratory or epidemiological studies. Lead and inorganic lead-based compounds have been classified by the International Agency for Research on Cancer (IARC) as possible human carcinogens (Group 2B), based on sufficient evidence of carcinogenicity in animals (IARC, 1987). Lead has also been classified as an A3 carcinogen (confirmed animal carcinogen with unknown relevance to humans) by the American Conference of Governmental Industrial Hygienists (ACGIH, 1998). The U.S. EPA has given lead a weight-of-evidence classification of B2, indicating lead is a probable human carcinogen and a confirmed animal carcinogen (IRIS, 1999). There is currently no established cancer slope factor for lead, which could be used to estimate cancer risk from an exposure amount.

4.1.6 Environmental Regulations for Lead

Apart from the regulations and recommendations regarding worker safety presented in the previous section, lead is regulated in a number of ways. This section presents a brief summary of the U.S. regulations for lead and lead compounds expected to impact facilities that manufacture materials for the computer display. It should be noted that many of the parts and materials which go into the manufacture of computer displays are manufactured in countries outside the U.S., with their own lead regulations which may differ significantly from those discussed below.

Air emissions of lead are regulated under the Clean Air Act (CAA) of 1970 and the amendments to the CAA of 1977 and 1990. Under the CAA, lead is regulated as a hazardous air pollutant (HAP), which is by definition a chemical that is generally known or suspected to cause serious health problems. Stationary source categories involved in the life cycle of a computer display that must meet new source performance standards include primary and secondary lead smelters, glass manufacturing plants, and metallic mineral processing plants (EPA, 1977; EPA, 1980a; ATSDR, 1999). A National Ambient Air Quality Standard (NAAQS) was also established for lead, requiring that the concentration of lead in air that the public breathes be no higher than 1.5 μ g/m³ averaged over 3 months [40 CFR 50.12].

Lead releases to surface water are regulated under the Clean Water and Effluent Guidelines and Standards promulgated under the Clean Water Act of 1977. Lead is identified as a priority pollutant [40 CFR 401.15], requiring the limitation of lead concentrations in pollutant discharges from point sources. The regulations also set standards of performance for new point sources, as well as pretreatment standards for both new and established sources. Regulated point source categories include lead smelters, steam electric power generation, glass manufacturers, and aluminum production and others, all of which contribute to the life-cycle impacts of a computer display. New point sources of lead contamination must also apply for National Pollution Discharge Elimination System (NPDES) permits which will establish effluent limits for sources of lead discharge.

To protect the population from a contaminated water supply, toxic substances in drinking water are regulated under the Safe Drinking Water Act of 1986. A federal drinking water standard of 15 ug/L has been established for lead.
EPA also regulates lead content in hazardous and solid wastes under the Resource Conservation and Recovery Act (RCRA). A solid waste containing lead or lead compounds may be considered a D008 characteristic hazardous waste if, when subjected to a Toxicity Characteristic Leachate Procedure (TCLP) test, the extract exceeds 5.0 mg/L [40 CFR 261.24] for lead. Other leadcontaminated wastes may be considered hazardous if specifically listed in 40 CFR 261.30-33, unless specifically excluded. Listed wastes from specific sources which contribute to the manufacture of computer displays include emission control dust from steel production and from lead smelting (K061 and K069 respectively), waste leaching solution of control dust from secondary lead smelting (K100), and spent baths and residues from electroplating operations containing cyanide (F006), which are sometimes used in PWB manufacturing. Specific sources of hazardous wastes, whether characteristic or listed wastes, are subject to handling, storage, and disposal restrictions detailed in the code of federal regulations.

Manufacturers who emit lead are required to report the quantity of the emissions under the Community Right-to-Know Act. EPA has recently reduced the reportable quantity threshold for lead from 10,000 lbs per year to 100 lbs per year of lead.

4.1.7 Alternatives to Lead Use in Computer Displays

Because of increasing pressure through regulation and market forces, attempts to reduce or eliminate lead in electronics have become popular. Several countries are considering or have already passed restrictions on the use and disposal of lead, prompting many companies to establish aggressive timelines for reducing or eliminating lead in their products. Several opportunities to eliminate or reduce the amount of lead used in a computer display are being aggressively researched. Two options being researched extensively are the development of a reduced lead frit, and lead-free solders for PWB manufacturing and assembly.

Although not large in mass in a monitor, frit glass is 70 to 80% lead by weight. Lead is one component of a mixture that crystalizes under intense heat, providing strength to the vacuum-tight frit seal. An alternative lead-free frit glass has been developed that is based on tin and zinc oxides, along with phosphate (Busio and Steigelmann, 2000). The lead-free glass is inherently mechanically weak, requiring large amounts of ceramic fillers (Al2O3) to be added to improve the mechanical strength of the seal. It also requires the addition of vitreous silica particles to match the thermal expansion requirements of the CRT glass. The resulting mixture requires a firing cycle approaching 450°C, which is typical of frit glasses. A drawback is that the frit glasses stay vitreous during the typical 30-60 minute furnace dwell time. However, initial evidence suggests that the fired frit seal remains rigid during the pumping step, which occurs at 350°C. Although comprehensive test results are not yet available, the high-temperature stability and rigidity of the lead-free frit glass is currently being tested under vacuum (Busio and Steigelmann, 2000).

Lead-free solders have been the subject of industry research for some time. Driven by renewed regulatory attention and by recent corporate commitments to reduce or eliminate lead from their product lines, alternatives to lead-based solder are garnering increased attention. Alternative lead-free solders include tin in combination with one or more of the following metals: silver, copper, bismuth, germanium, and antimony. Several companies, including Sony, Toshiba, Hitachi, and Ford Motor Company, have either already begun to implement electronics production using a lead-free solder alternative, or have announced plans to do so. Though still a

relatively new and untested technology, initial testing has shown that alternatives are capable of producing quality component connections, though they have a narrower operating window and require higher temperatures to apply (Keenan and Kellett, 2001).

4.2 MERCURY

Mercury is not only used in the manufacturing of LCDs, but also is emitted from a number of processes over the life cycle of both LCDs and CRTs. Because of mercury's toxicity to both humans and the environment, this section presents a more detailed look at the uses and impacts of mercury, and its potential for causing harm as a result of the manufacture, use, and disposal of computer displays.

4.2.1 Mercury in Computer Displays

Mercury is an important material in the construction of cold cathode fluorescent lamps (CCFLs) that are used to backlight the LCD. A typical LCD utilizes a minimum of two CCFLs, and can use as many as eight in larger displays. The CCFL consists of a glass tube filled with a small amount of rare gas (typically argon) and a few drops of mercury. Metal electrodes built into the ends of the tube conduct electric current to the inside gas, vaporizing a portion of the mercury which then becomes excited, emitting light in the ultraviolet spectrum. Fluorescent phosphors, which coat the inside of the glass tube, convert the ultraviolet emissions from the mercury gas into visible white light. The phosphors are responsible for nearly all of the visible light from the lamp, with the visible mercury spectrum contributing only a little to the lamps output (Srivastava and Sommerer, 1998). No mercury is contained directly within the CRT.

Although a great deal of mercury is obtained by CCFLs, there also is a small source of mercury generated by the infrequent breakage of mercury lamps in the photolithographic exposure systems used to make both the CRT and LCD. Mercury filters are used in some water cooled exposure table equipment to catch the mercury from lamp explosions or trap it in water cooling baths. In air cooled lamp usage, the proximity of aluminum metal gives sites for amalgam formation and a number of clean-up procedures are used. Unbroken lamps are returned to the lamp manufacturer for recycling or disposal (Donofrio and Eckel, 1999).

Mercury may also be emitted from several processes required to manufacture, operate, and dispose of both CRTs and LCDs. For example, electricity generated from the combustion of coal results in the emission of mercury contained within the fuel. Other processes potentially responsible for mercury emissions include mercury ore processing, non-ferrous metal production, and the recycling of LCDs at the end of their useful life. The amount of mercury released through these incidental mercury emissions is comparable to the amount of mercury used as a direct material in the manufacture of LCD backlights.

4.2.2 Life-Cycle Inputs and Emissions of Mercury for Computer Displays

Data on mercury and mercury-containing materials were collected and compiled as part of the life-cycle inventory. Material inputs containing mercury include primary materials (e.g., CCFLs) which end up as part of the product, as well as ancillary materials (e.g., fossil fuels containing mercury) which are consumed as part of the manufacturing process or other supporting processes (e.g., energy production). The data were aggregated by material from individual processes and are presented by life-cycle stage for LCDs in Table 4-11. More detailed material input data, which include the processes that use each input and the quantity of mercury releases, are included in Appendix N.

100010		JPSI		i
Life-cycle stage	Inputs	Quantity	Units	Туре
Manufacturing	Mercury	3.99e-06	kg	Primary material
Manufacturing	Backlight lamp (CCFL)	1.94E-03ª	kg	Product

Table 4-11. Life-cycle stage mercury inputs for LCDs

^a Quantity of release shown represents entire mass of input material. Mercury may only comprise a fraction of the total mass of material shown.

Mercury is not listed as an input in the life-cycle inventory for CRTs. However, mercury is contained in raw material inputs (e.g., mercury contaminants in fossil fuels burned to produce energy) into processes such as non-ferrous metal production and energy production for both LCDs and CRTs. Though mercury inputs exist, they occur in upstream manufacturing processes where input data did not contain detailed composition data for fuel inputs.

Releases of mercury and mercury-containing materials into the environment occur throughout the entire life cycle of the computer display. Environmental releases include airborne, waterborne, solid waste, and hazardous waste emissions. Similar to the inputs, emissions data were aggregated by the material released from individual processes and then reported by life-cycle stage. The mercury and mercury-containing material released, the quantity of the release, the type of release (e.g., waterborne), and the ultimate disposition of the release all affect the nature and type of environmental impacts.

The total life-cycle outputs/emissions containing mercury for both CRTs and LCDs are organized by output type and shown in Tables 4-12 and 4-13, respectively. More detailed data on mercury and mercury-containing outputs for each process are presented in Appendix N.

		0	-		
Life-cycle stage	Outputs	Quantity	Units	Туре	Disposition
Materials processing	Mercury	3.00E-06	kg	Airborne	Air
Materials processing	Mercury	1.42E-10 ^a	kg	Solid waste	Landfill
Materials processing	Mercury compounds	9.68E-07	kg	Waterborne	Surface Water
Manufacturing	Mercury	1.12E-06	kg	Airborne	Air
Manufacturing	Mercury compounds	1.35E-12	kg	Waterborne	Surface Water
Use	Mercury	7.51E-06	kg	Airborne	Air
End-of-life	Mercury	-1.15E-07	kg	Airborne	Air
End-of-life	Mercury compounds	4.33E-11	kg	Waterborne	Surface Water

 Table 4-12. Life-cycle stage mercury outputs from CRTs

^a Quantity of release shown represents entire mass of waste disposed. Mercury may only comprise a fraction of the total mass of waste shown.

-				-	-
Life-cycle stage	Outputs	Quantity	Units	Туре	Disposition
Materials processing	Mercury	9.44E-07	kg	Airborne	Air
Materials processing	Mercury compounds	5.82E-07	kg	Waterborne	Surface Water
Manufacturing	Broken CCFL	2.69E-07 ^a	kg	Solid waste	Landfill
Manufacturing	Mercury	2.64E-06	kg	Waterborne	Treatment
Manufacturing	Mercury compounds	6.52E-14	kg	Waterborne	Surface Water
Manufacturing	Waste CCFL, with mercury	8.17E-10 ^a	kg	Hazardous waste	Treatment
Manufacturing	Waste glass, with mercury	1.05E-10 ^a	kg	Hazardous waste	Landfill
Manufacturing	Wastewater stream, from CCFL mfg.	167	kg	Waterborne	Treatment
Use	Mercury	2.80E-06	kg	Airborne	Air
End-of-life	Mercury	-8.64E-08	kg	Airborne	Air
End-of-life	Mercury compounds	1.62E-11	ka	Waterborne	Surface Water

Table 4-13. Life-cycle stage mercury outputs from LCDs

^a Quantity of release shown for solid waste and hazardous waste represents entire mass of waste disposed. Mercury may only comprise a fraction of the total mass of waste shown.

Mercury is released into the environment in many forms, but is most typically an airborne release. The largest air emissions of mercury result from the generation of electricity from fossil fuel burning. For LCDs, there is nearly the same amount of mercury emitted to the air from energy production (3.22 mg) as the mass of mercury used in the fabrication of an LCD (3.99 mg). In fact, the amount of mercury emitted to the air from electricity generation for CRTs (7.75 mg) is greater than the entire amount of mercury from both the fabrication and energy production for an LCD. Other airborne releases include the processing of ores such as lead, and the production of several raw materials such as aluminum, polycarbonate, and steel.

4.2.3 Computer Display Life-Cycle Impacts for Mercury

The life-cycle impacts of mercury, mercury-based compounds, and materials containing mercury (e.g., waste glass from broken CCFLs) calculated for CRTs and LCDs during the LCIA are summarized in Tables 4-14 and 4-15, respectively. Impact scores in the table are expressed in units specific to each impact category (see Chapter 3.1 for a discussion of impact category units and weighting). The total impact score for each category resulting from mercury and mercury-based materials is presented at the bottom of each table.

				Impact scores	s by category		
Life-cycle stage	Material	Hazardous waste landfill use (m ³)	Solid waste landfill use (m ³)	Chronic health effects- public (tox-kg)	Chronic health effects- occupational (tox-kg)	Aquatic toxicity (tox-kg)	Terrestrial toxicity (tox-kg)
Materials processing	Mercury	0	0	3.00E-06	0	0	3.00E-06
Materials processing	Mercury compounds	0	0	5.11E-04	0	9.02E-04	5.10E-04
Manufacturing	Mercury	0	0	1.12E-06	0	0	1.12E-06
Manufacturing	Mercury compounds	0	0	7.13E-10	0	1.26E-09	7.11E-10
Use	Mercury	0	0	7.51E-06	0	0	7.51E-06
End-of-life	Mercury	0	0	-1.15E-07	0	0	-1.15E-07
End-of-life	Mercury compounds	0	0	2.29E-08	0	4.04E-08	2.28E-08
Total Impact Scores	by Category	0	0	5.22E-04	0	9.02E-04	5.21E-04

Table 4-14. Summary of mercury-based impact scores by impact category for CRTs

				Impact sco	Impact scores by category			
Life-cycle stage	Material	Hazardous waste landfill use ^a (m ³)	Solid waste landfill use ^a (m ³)	Chronic health effects-public (tox-kg)	Chronic health effects- occupational (tox-kg)	Aquatic toxicity (tox-kg)	Terrestrial toxicity (tox-kg)	
Materials processing	Mercury	0	0	9.44E-07	0	1.82E-07	9.44E-07	
Materials processing	Mercury compounds	0	0	3.07E-04	0	5.42E-04	3.07E-04	
Manufacturing	Broken CCFL	0	1.98E-11	0	0	0	0	
Manufacturing	Mercury	0	0	5.54E-07	3.99E-06	1.9387E-07	5.54E-07	
Manufacturing	Mercury compounds	0	0	3.44E-11	0	6.08E-11	3.44E-11	
Manufacturing	Waste glass, with mercury	7.73E-15	0	0	0	0	0	
Use	Mercury	0	0	2.80E-06	0	0	2.80E-06	
End of Life	Mercury	0	0	-8.64E-08	0	0	-8.64E-08	
End of Life	Mercury compounds	0	0	8.53E-09	0	1.51E-08	8.52E-09	
Total Impact Sc	cores by Category	7.73E-15	1.98E-11	3.11E-04	3.99E-06	5.43E-04	3.11E-04	

	-				
Table 4 15	Summany	f manany had	d impost coores	hy impost	antogomy for I CDc
1 able 4-15.	Summary o	i mercurv-Dase	u mndaci scores	DV IIIDACL	category for LCDS

^a Percentages of impacts shown for solid and hazardous wastes are based on the entire mass of material disposed, not necessarily on the amount of mercury. As such, the percentage over-estimates the impact of mercury to either the solid or hazardous waste landfill use.

4.2 MERCURY

Impact scores for some mercury-based inputs and outputs shown in Tables 4-11 through 4-13 were not calculated if the type and disposition of the input or release was not expected to contribute to any of the impact categories. For example, a waterborne release of mercury with a disposition going to treatment assumes that the mercury was treated prior to being released to the environment. However, since inventory data for subsequent treatment/disposal processes could not be obtained, it was assumed the mercury (or other inventory item) had been removed to a level such that the subsequent release of treated wastewater would not contribute significantly to aquatic toxicity impacts. Similarly, impact scores were not calculated for releases going to recycling/reuse or for those designated as a product.

The life-cycle mercury-based outputs from LCDs had a broader affect on the environment than those from CRTs, impacting a wider group of impact categories. Impacts to both solid and hazardous waste landfill use, as well as to the chronic health effects of workers, all directly result from the use of mercury in the LCD backlights. No mercury is required for the fabrication of a CRT. Although the quantities are not large (see Tables 4-11 through 4-13), they cannot be discounted, given the toxicity of mercury to both human health and the environment.

Chronic occupational toxicity impacts were only calculated for mercury inputs to processes in the CDP. The overall impact scores (3.99E-06 tox-kg for LCD, none for CRT) likely underestimate the chronic occupational impacts for mercury, because they are based on inputs only and do not consider chronic occupational impacts from outputs in other processes such as aluminum production or fluorescent lamp recycling, which may result in emissions of mercury that originate within the workplace.

The contribution of mercury-based impacts for each computer display technology to the overall impacts for each individual impact category is shown in Table 4-16. Values in the table are expressed in the percent contribution the material made to the overall impact score for all materials (e.g., liquid crystals, fuel oil, glass) for each category. The percent contributions give an indication of the importance of mercury-based impacts relative to the life-cycle impacts from other materials or outputs from the computer display.

Impact category	CRT	LCD
Hazardous waste landfill use	N/A	NA
Solid waste landfill use	N/A	3.65E-08
Chronic health effects- public	2.64E-05	3.45E-05
Chronic health effects- occupational	N/A	5.80E-07
Aquatic toxicity	4.01E-01	1.05E-02
Terrestrial toxicity	2.64E-05	3.48E-05

 Table 4-16. Summary of percent contributions from mercury-based materials to individual impact categories

N/A = Not applicable

The results from Table 4-14 and 4-15 indicate that the mercury impacts from a CRT exceed the impacts from an LCD in categories common to both technologies. This was not expected, because mercury is used intentionally in an LCD, but not in a CRT. However, the results are not surprising because mercury emissions from coal-fired power plants are known to be one of the largest anthropogenic sources of mercury in the United States. Because the CRT consumes significantly more electricity in the use stage than the LCD, its use stage emissions of mercury are proportionately higher than those of the LCD. In fact, the mercury emissions from the LCD, including both the mercury used in LCD backlights and the mercury emissions from electricity generation in the use stage that can be attributed to the LCD.

The impacts resulting from mercury and mercury-based materials do not appear to be significant relative to the total impacts of all of the computer display materials (e.g., liquid crystals, lead solder), as shown in Table 4-16. The largest contribution is 0.4% of the total aquatic toxicity impacts for CRTs, and 0.01% of the total aquatic toxicity impacts for LCDs. Impacts to other categories from both LCDs and CRTs were minimal.

4.2.4 Exposure Summary

Mercury may pose a threat to human health anytime there is the potential for human exposure throughout the life cycle of a computer display. Exposure occurs anytime a chemical or physical agent, in this case mercury or mercury compounds, come into contact with an organism, be it human or ecological. This section qualitatively identifies potential exposures for workers in facilities using mercury (occupational exposures), the general public, who may be exposed to mercury releases into the ambient environment, and the ecological population.

4.2.4.1 Occupational exposures

About 4 mg of elemental mercury (combined total of all mercury contained within the backlights) is used to manufacture the fluorescent backlight for the LCD backlight unit assembly. Workers manufacturing the backlights may be exposed to the mercury used for these lights. This study found no information on the specific manufacturing processes that are used to make CCFLs or the specific worker exposures that could occur, but did find manufacturing information for generic fluorescent lamps. We assume the processes are similar and have included a brief discussion of the fluorescent lamp manufacturing process and potential sources of worker exposure, below.

In fluorescent lamp manufacturing, pre-cut bulbs are washed, dried, and coated with a liquid phosphor emulsion that deposits a film on the inside of the bulb. Mount assemblies are then fused to each end of the bulb and the bulb is transferred to an exhaust machine. The bulb is then exhausted and mercury is injected into the bulb. Some of the mercury combines with the emulsion on the interior of the bulb where it remains over the life of the bulb. The glass bulb is then filled with an inert gas and sealed (EPA, 1997).

During the lamp manufacturing process, emissions of mercury can occur from transfer and parts repair during mercury handling, by the mercury injection operation, and from broken lamps, spills, and waste materials. Workers can be exposed to mercury from any of these sources, but mercury air levels can be reduced by process modifications, containment, ventilated inclosures, local exhaust ventilation, and temperature control (EPA, 1997).

4.2 MERCURY

There were no other inputs of mercury reported in the LCD life-cycle inventory, and none were reported in the CRT life cycle. Workers may also be exposed to inorganic mercury in the fluorescent backlights when processing LCDs at the end-of-life. Family members of workers may be exposed as well, from a worker's clothing or shoes if they are contaminated with mercury and brought home.

The processes associated with LCI mercury inputs and releases are presented in supplemental tables in Appendix N. It is important to note that while this list gives an indication of where likely mercury exposures may occur, it is not exhaustive. Many processes and subprocesses may be contained within a process listed, each of which may pose its own potential for occupational mercury exposures.

4.2.4.2 General population exposures

Mercury is a persistent biaccumulative toxic (PBT) chemical, as designated by EPA (EPA, 2001b). PBT pollutants are highly toxic, long-lasting substances that can build-up in the food chain to levels that are harmful to human and ecosystem health. The general public may be exposed to metallic mercury that is not safely contained (although it is unlikely that the backlight would break to release mercury during normal LCD use) or to methylmercury-contaminated foods. Mercury also can be passed from a pregnant woman to developing child through the placenta, and from a mother to nursing infant through breast milk.

Most of the mercury released to the environment throughout the CRT and LCD life cycles results from electricity generation required for use, manufacturing, and materials processing life-cycle stages. A total of approximately 4 mg of mercury and mercury compounds are released to air for an LCD, and approximately 12 mg are released to air for a CRT. Mercury is naturally present in coal and becomes airborne when coal is burned to generate electricity. Airborne mercury can stay in the atmosphere for up to a year, and can travel thousands of miles (EPA, 2001a). EPA modeling suggests that "a substantial fraction" of the mercury released to air by utilities is dispersed "well beyond the local area" due to the fine particulate nature of the emissions and tall stacks (EPA, 1998). Mercury in the atmosphere moves to land and water by settling out with particles and being washed out by rain (dry and wet deposition). It may be deposited directly to water or be carried by runoff to a lake, stream, or ocean. In the LCIs, in addition to the air releases of mercury, 0.6 mg of mercury and mercury compounds are released directly to surface water in the LCD life cycle, and 1 mg in the life cycle of the CRT. Ultimately, at the end of life, the 4 mg of mercury in the LCD backlight will most likely be released to the environment during LCD recycling or disposal processes.

Surface water is the environmental medium of most concern for mercury. In a surface water environment, inorganic mercury can be transformed into methylmercury, a form which readily bioaccumulates in fish (inorganic mercury does not tend to bioaccumulate). An EPA study of mercury supports a "plausible link" between releases of mercury from industrial and combustion sources and methylmercury found in fish³ (EPA, 2001c). Methylmercury

³ The mercury released to the environment from the CRT and LCD life cycles are only part of the overall burden of mercury released to the environment from coal-fired power plants for all uses of electricity. The proportion of mercury in fish that is due to coal-fired power generation is not known. In addition, there are other natural and anthropogenic sources of mercury to the environment.

concentrations at the top of the food chain (such as in predatory fish or fish-eating animals) can be thousands or even millions of times greater than that in the surface water itself (EPA, 2000a).

The most important mercury exposures to the general public result from eating fish that are contaminated with methylmercury. The populations of most concern are children and women of child-bearing age (the developing fetus may be the most sensitive to the effects of methylmercury). Also of concern are people whose diet largely depends on fish, such as with some native cultures. The overall amount of exposure to mercury from eating fish depends on both the concentration of mercury in the fish, and on the amount of fish a person regularly eats. Fish advisories due to methylmercury contamination have been issued by EPA⁴, 39 states, and some tribes, providing consumption limits for certain species of fish (EPA, 2000a, 2001c).

Freshwater fish are most affected, but some saltwater fish have also been found to be contaminated with methylmercury. The Food and Drug Administration (FDA) has issued a fish consumption advisory for pregnant women, nursing mothers, and small children to avoid eating certain large saltwater fish (shark, swordfish, king mackerel, and tilefish), and to limit overall weekly fish consumption, due to methylmercury contamination (FDA, 2001).

4.2.4.3 Exposure and effects to ecological populations

In addition to the fish themselves being contaminated, wildlife that eat fish also may be at risk from exposure to methylmercury. Species of concern include loons, eagles, mink, otter, wood stork, and the endangered Florida panther. Adverse effects of mercury to wildlife include death, reproduced reproductive success, impaired growth and development, and behavioral abnormalities. Levels have been measured in some individual wild animals that are comparable to those levels seen to cause harmful effects in laboratory tests with the same species (EPA, 2001c). Ambient water criteria have been developed by EPA under the CWA. Criteria for the protection of aquatic life for mercury are presented in Table 4-17.

⁴ "EPA is issuing a national advisory concerning risks associated with mercury in freshwater fish caught by friends and family. The groups most vulnerable to the effects of mercury pollution include: women who are pregnant or may become pregnant, nursing mothers, and young children. To protect against the risks of mercury in fish caught in fresh waters, EPA is recommending that these groups limit fish consumption to one meal per week for adults (6 ounces of cooked fish, 8 ounces of uncooked fish) and one meal per week for young children (2 ounces cooked fish or 3 ounces of uncooked fish)." – EPA Office of Water, January 2001

Type of criteria	Criteria value (µg/L)	Notes					
For protection of aquatic life							
Freshwater Criteria Maximum Concentration (CMC)	1.4 ^a	the acute limit for the priority pollutant in freshwater					
Freshwater Criterion Continuous Concentration (CCC)	0.77 ^a	the chronic limit for the priority pollutant in freshwater					
Saltwater Criteria Maximum Concentration (CMC)	1.8 ^a	the acute limit for the priority pollutant in saltwater					
Saltwater Criterion Continuous Concentration (CCC)	0.94 ª	the chronic limit for the priority pollutant in saltwater					

Table 4-17. EPA water quality criteria for mercury

Source: Water Quality Criteria and Standards, EPA Office of Water. Revised: 06/21/2001 http://oaspub.epa.gov/wqsdatabase/epa.rep_parameter; report for mercury

^a Criteria for metals are expressed in terms of the dissolved metal in the water column. This recommended water quality criterion was derived from data for inorganic mercury (II), but is applied here to total mercury. If a substantial portion of the mercury in the water column is methylmercury, this criterion will probably be underprotective (EPA is updating the ambient water quality criteria on methylmercury.) In addition, even though inorganic mercury is converted to methylmercury and methylmercury bioaccumulates to a great extent, this criterion does not account for uptake via the food chain because sufficient data were not available when the criterion was derived.

4.2.5 Human Health Effects

Mercury affects the nervous system, brain, and kidneys. Effects on the nervous system vary depending on the form of mercury. Inorganic mercury salts, for instance, do not enter the brain as readily as elemental mercury or methylmercury. Symptoms of mercury effecting the brain and nervous system include personality changes, tremors, changes in vision such as narrowing of the visual field, deafness, loss of muscle coordination, loss of sensation, and problems with memory (ATSDR, 1999).

The fetus, infants, and young children are especially susceptible to the effects of mercury on the nervous system. As mentioned above, mercury (especially methylmercury in food) can be passed from a pregnant woman to the unborn developing child, and from a mother to a nursing infant through breast milk. The developmental effects of mercury vary in severity depending on the amount of exposure. Children exposed in this way may show small decreases in IQ or may be slower to walk and talk. More severe effects might include brain damage with mental retardation, blindness, muscle weakness or seizures, and inability to speak (ATSDR, 1999).

Mercury accumulates in the kidneys and all forms of mercury can cause kidney damage at higher exposures (ATSDR, 1999). Short term exposure (hours) to high levels of mercury vapor, as might occur through an accidental spill in the workplace, include damage to the lining of the mouth, irritated lungs and airways, nausea, vomiting, diarrhea, increased blood pressure or heart rate, skin rashes, and eye irritation. Skin contact may cause an allergic reaction (skin rashes) in some people (ATSDR, 1999).

4.2.5.1 Chronic effects (noncancer)

A reference dose (RfD) is an estimate (with uncertainty spanning perhaps an order of magnitude) of the daily exposure through ingestion to the human population (including sensitive subgroups) that is likely to be without an appreciable risk of deleterious noncancer effects during a lifetime (in mg/kg-day). Similarly, a reference concentration (RfC) is an estimate (with uncertainty spanning perhaps an order of magnitude) of the daily inhalation exposure to the human population (including sensitive subgroups) that is likely to be without an appreciable risk of deleterious noncancer effects during a lifetime (in mg/m³) (Barnes and Dourson, 1988). RfDs and RfCs established by EPA for mercury and mercury compounds are presented in Table 4-18.

Form of mercury ^a	Ingestion: reference dose (RfD) (mg/kg- day) ^b	Inhalation: reference concentration (RfC) (mg/m ³) ^c	Notes, Source
Elemental mercury (Hg)	not available	0.0003	Based on hand tremor, memory disturbances, and other effects seen in human occupational inhalation studies (IRIS, 2001).
Methylmercury (CH ₃ Hg ⁺)	0.0001	not available	Based on developmental neuropsychological impairment seen in human epidemiological studies (IRIS, 2001).
Mercuric chloride (HgCl)	0.0003	not available	Based on autoimmune effects in rats (IRIS, 2001).

Table 4-18. Chronic toxicity reference values for mercury and mercury compounds

^a Forms of mercury for which EPA has established an RfD or RfC

^b milligrams per kilogram of body weight per day for oral exposure (ingestion)

^c milligrams per cubic meter of air, assuming continuous inhalation exposure for a 70-kg adult

4.2.5.2 Carcinogenicity

EPA has determined that mercury chloride and methylmercury are Possible Human Carcinogens (cancer weight of evidence [WOE] classification C) based on limited evidence of carcinogenicity in animals and inadequate or lack of human data. Elemental mercury is classified by EPA as Not Classifiable as to Human Carcinogenicity (WOE class D) based on inadequate or no evidence of carcinogenicity (IRIS, 2001).

4.2.6 Environmental Regulations for Mercury

This section presents a brief summary of the U.S. regulations for mercury and mercury compounds that may affect facilities that manufacture materials for the computer display or are otherwise affected by the life cycle of computer displays. It should be noted that many of the parts that go into the computer display are manufactured in other countries with their own regulations which may differ significantly from those presented below.

Air emissions of mercury are regulated under the CAA of 1970 and the amendments of 1977 and 1990. Under the CAA, mercury is regulated as a hazardous air pollutant (HAP), which is by definition a chemical that is either known or is suspected to cause serious health problems for humans. EPA established National Emission Standards for HAPs (NESHAPs) for mercury emissions based on risk under the pre-1990 version of the Clean Air Act. These NESHAPS [40 CFR 61 Subpart E] cover three source categories: ore processing facilities, mercury cell chlor-alkali plants, and sewage sludge driers. Specific source requirements are specified for, among other things, municipal waste combusters, hazardous waste combusters, and mercury ore processing facilities. These source requirements could take the form of either a NESHAP or a maximum achievable control technology (MACT) requirement.

OSHA has established standards for protecting worker health through the maintenance of a safe working environment. The OSHA permissible exposure limit (PEL) for workplace exposure to mercury is 0.1 mg/m³ (8-hour time weighted average). NIOSH has also established recommended exposure limits (RELs) for several mercury compounds including a REL of 0.05 mg/m³ for mercury vapor.

Mercury emissions to surface water are regulated under the CWA, which lists mercury as a priority pollutant [40CFR 401.15], requiring the limitation of mercury in point source discharges. For mercury discharges, CWA regulations specify technology-based effluent limits for classes and categories of industries (see 40 CFR 401, 403, Appendix B), and describe the rights of states to establish effluent limits more stringent than technology-based standards. Technology-based standards are listed for the following specific industries and point sources involved in computer display manufacturing: nonferrous metals production, including primary precious metals and mercury (40 CFR 250); secondary mercury (40 CFR 421.200); steam electric power generation (40 CFR 423- Appendix A); and mercury ore mining (40 CFR 440.40). The CWA also requires that new and existing points sources of mercury obtain a NPDES permit, which will establish effluent limits for mercury discharges to surface waters.

To protect human health and preserve the nations drinking water supply, EPA has been tasked by the SDWA to establish safe drinking water standards for toxic chemicals. In accordance with the SDWA, EPA has established a safe drinking water standard for mercury of $2 \mu g/L$.

The release or disposal of solid or hazardous waste containing mercury is regulated under RCRA, which outlines specific classification and disposal requirements for products and wastes that contain mercury. Mercury is both a characteristic and a listed waste under RCRA. A solid waste containing mercury may be considered a D009 characteristic hazardous waste if, when subjected to a TCLP test, the extract exceeds 0.2 mg/L [40 CFR 261.24] for mercury. Other mercury-contaminated wastes may be considered hazardous if they are specifically listed in 40 CFR 261.30-33, unless they are specifically excluded. Listed wastes for mercury include leachate resulting from the disposal of more than one restricted waste classified as hazardous (F039), and wastewater treatment sludge and brine purification muds resulting from the mercury

cell process in chlorine production (K106 and K071 respectively). Hazardous wastes are subject to land disposal restrictions requiring that wastes be treated to below regulatory threshold levels before they may be land-disposed.

In order to reduce the amount of hazardous waste in a landfill, EPA established the Universal Waste Rule (UWR) in 1995. The rule was intended to encourage the recycling and proper disposal of common hazardous waste components found in municipal waste streams, and reduce the regulatory burden on businesses who produce these wastes. The rule allows for less stringent standards for storing, transporting, and collecting wastes. However, the waste must comply with full hazardous waste requirements for final recycling, treatment, or disposal. Batteries and fluorescent lamps are included in the rule.

EPA also has regulated the air emissions of mercury from hazardous waste combustion and from industrial boilers and furnaces under RCRA. EPA has issued new standards for mercury emissions from these sources.

In December 2000, EPA announced plans to require coal-fired power plants to cut their emissions of mercury. EPA plans to propose the regulations by 2003, with final rules in place by 2004 (EPA, 2000a). More recently, in April 2001 the Bush administration sought to dismiss an electric industry lawsuit that would stop the EPA from regulating mercury and other toxic air pollutants (Doggett, 2001).

Manufacturer's who emit mercury are required to report the quantity of the emissions under the Emergency Planning and Community Right-to Know-Act (EPCRA). EPA has established a reportable quantity threshold of 10 pounds for a facility that manufactures, processes, or otherwise uses mercury. A similar threshold of 10 pounds exists for facilities that manufacture, process, or otherwise use mercury compounds. Data that has been reported by facilities for mercury is made available to the public through the publication of the Toxics Release Inventory (TRI).

4.2.7 Alternatives to Mercury Use in LCDs

In an effort to minimize or even eliminate the quantity of mercury used during the fabrication of LCDs, manufacturers have begun to develop mercury-free alternatives to mercury vapor backlights. One such alternative is a flat lamp which is filled with inert gas xenon in place of the typical mercury vapor lamps. The lamp has the appearance of a large white tile about one centimeter in thickness. Its dimensions are the exact same as the screen itself, illuminating the image evenly, hence eliminating the need for complex optical systems to distribute the light. The new lamp is capable of emitting enough light to make the monitor twice as bright, making it possible to use the screen during daylight, while also extending the viewing angle (OSRAM, 1998).

The new lamps generate light in a fashion similar to conventional backlight lamps. An electrical current passed through a gas discharge produces ultra-violet light which is then converted to visible light by phosphors. Unlike mercury gas which causes 'greying' in the phosphors over time, the xenon gas does not affect the phosphors, extending the life of the lamp. The lights have an average lifetime of up to 50,000 hours compared to only about 20,000 hours for the conventional mercury backlights, extending the life of computer LCD displays before they have to be replaced (OSRAM, 1998).

A drawback of the lamps, and the current subject of research is the reduced energy efficiency of the new mercury-free lamps. The luminous efficiency of the new mercury-free

lamp is only about half of the efficiency of a conventional backlight, due to less efficient conversion of the UV light to visible light by the phosphors (OSRAM, 1998).

4.3 LIQUID CRYSTALS

One of the most significant differences between the two computer displays is the use of liquid crystals in the LCD to generate an image. The toxicity of liquid crystals in LCDs has been alluded to in literature indicating the potential for human health concerns. Because of relative lack of information about these compounds, this section provides a more detailed look a liquid crystals to better understand their potential impacts on human and ecological populations,

4.3.1 Liquid Crystals in Computer Displays

Liquid crystals (LCs) are organic compounds with the optical and structural properties of crystals, but with the mechanical features of fluids. There are hundreds of LC compounds that may be used in an LCD, each with different physical and optical characteristics. They are typically classified by molecular weight, with low molecular weight LCs typically used for LCD computer displays. LCs are not required for the fabrication of CRTs.

LCs are responsible for forming and transmitting the image produced by an LCD. The LC portion of an LCD typically consists of as many as 20 different LC substances, mixed together to form a white, opaque liquid that flows easily (Merck, 1999). The mixture consists of elongated molecules that are held together at their ends by polar forces and aligned in the same direction. The molecules move together in a series of flexible molecular chains, with each chain influencing the alignment of other chains. By exposing the molecular chains to electric fields, the alignment of the chains, and by extension their ability to transmit light, can be manipulated (SEMI, 1995).

LCDs are fabricated using a complex multi-step process in a clean room (refer to Chapter 1 for a more detailed description of the fabrication process). Once the front and back layers have been manufactured and assembled into a display cell, the LC is ready to be added. The assembled display cells are placed into a vacuum chamber containing a reservoir of the LC mixture, and the chamber is evacuated. A corner of the empty display cell is lowered into the LC mixture using a remote control. Nitrogen gas is then introduced into the evacuated chamber to bring the pressure up to approximately 1 atmosphere, exerting pressure on the surface of the LC mixture, forcing it up into the display cell (SEMI, 1995). Approximately 0.6 mg of LC is required for every square centimeter of panel surface (Merck, 1999). The display cell is sealed once the LC fully penetrates the cell.

4.3.2 Life-Cycle Inputs and Outputs of Liquid Crystals for Computer Displays

Data on LC compounds were collected and compiled as part of the life-cycle inventory. The data were aggregated by material from individual processes and presented by life-cycle stage for LCDs in Table 4-19.

The individual LC compounds identified in Table 4-19 formed the ingredients of the liquid crystal mixture used in the LCD. These compounds, used in varying quantities, are mixed together in a liquid crystal manufacturing process to develop a mixture with the desired optical characteristics for the LCD. While the above inputs illustrate the quantities of individual LC compounds found in the LCD evaluated in this LCA, LC mixtures found in other LCDs could be comprised of up to 20 or more liquid crystal compounds selected from the hundreds of

4.3 LIQUID CRYSTALS

compounds currently available for use in LCDs. A small quantity of the liquid crystal mixture (1.2 grams) is then used in the fabrication of the LCD.

Life-cycle stage	Inputs ^a	Quantity	Units	Туре
Manufacturing	Liquid crystal A	0.26	g	Primary material
Manufacturing	Liquid crystal B	0.37	g	Primary material
Manufacturing	Liquid crystal C	0.22	g	Primary material
Manufacturing	Liquid crystal D	0.33	g	Primary material
Manufacturing	Liquid crystal E	0.070	g	Primary material
Manufacturing	Liquid crystal F	0.34	g	Primary material
Manufacturing	Liquid crystal G	0.19	g	Primary material
Manufacturing	Liquid crystal mixture, for 15" LCD	1.2	g	Primary material

 Table 4-19. Life-cycle stage liquid crystal inputs

^a Identities of liquid crystal compounds have been masked to protect the confidentiality of the compound names.

Life-cycle inventory data indicate that LCs are primarily released at the time of the product's final disposition. Although other outputs of liquid crystals surely exist, they are likely minimal and were not identified in the life-cycle inventory collected for LCDs. Evaporative LC emissions during the manufacturing and mixture formulation processes are minimal due to the typically low vapor pressures of LC compounds (Becker, 2001). Releases resulting from broken or defective LCDs during manufacture are also likely to be small because of the strong adhesives forces between LCs and the polymer film covering the outer layer of the displays (Becker, 2001). However, small amounts of LC compounds were observed to present in the waste of one LCD manufacturing facility during data collection (Overly, 2001). Because data were not provided from manufacturers on manufacturing releases, and no data were available for releases at end-of-life, it is difficult to definitively quantify the releases or their significance to the environment.

4.3.3 LCD Life-Cycle Impacts of Liquid Crystals

The life-cycle impacts of LC compounds calculated for LCDs during the LCIA are summarized in Tables 4-20. Impact values in the table are expressed in units specific to each impact category (see Chapter 3.1 for a discussion of impact category units and weighting). The total impact score for each category from LCs is presented at the bottom of the table.

Life-cycle stage	Material	Chronic health effects- occupational (tox-kg)
Manufacturing	Liquid crystal A	5.29E-04
Manufacturing	Liquid crystal B	4.35E-04
Manufacturing	Liquid crystal C	3.89E-04
Manufacturing	Liquid crystal D	1.40E-04
Manufacturing	Liquid crystal E	6.84E-04
Manufacturing	Liquid crystal F	6.53E-04
Manufacturing	Liquid crystal G	7.31E-04
Total category impac	ct score	3.56E-03

 Table 4-20. Summary of LCD liquid crystal impacts

Impact scores have been calculated based on the inventory item, release type, and its reported disposition. Occupational impacts to workers as a result of LC inputs are shown in the Table 4-20. Impacts were not calculated for LCs which end up as part of the product because users of the product are not expected to become exposed to the LC compounds during the typical operation of the LCD. In addition, because releases of LCs to the environment were not provided by the manufacturers in the LCI, potential impacts resulting from these releases were also not assessed in this project. LCs are not used to fabricate CRTs and so have no environmental impacts in the CRT life cycle.

LCs do not appear to contribute significantly to any of the impact categories defined for this study. The total score for occupational impacts based on potential worker exposure to LCs of 4.18 tox-grams represents less than 0.01% of the total overall chronic occupational health effects impact score of 898 tox-kg for the functional unit of one LCD.

4.3.4 Exposure Summary

Like any materials classified as potentially toxic, LCs have the potential to pose a threat to human health anytime there is the potential for human or ecological exposure throughout the life cycle of a LCD. Exposure occurs anytime a chemical or physical agent, in this case LC compounds, come into contact with an organism, be it human or ecological. This section qualitatively identifies potential exposures for workers in facilities using LCs (occupational exposures), the general public, who may be exposed to LC releases into the ambient environment, and the ecological population.

4.3.4.1 Occupational exposures

Occupational exposures to LCs during the fabrication of the LCD panels are not expected to be significant. LCD panels are fabricated in a clean room environment. The previously assembled but empty display cells are placed into a vacuum chamber containing a reservoir of the LC mixture, and lowered into the LC mixture using a remote control. Approximately 1.2 grams of LC compounds are used to develop the LCD panel. The enclosed nature of the chamber combined with the equipment (e.g., gloves, aprons) worn by workers in a clean room environment may both act to minimize exposures.

4.3 LIQUID CRYSTALS

However, the potential for other occupational exposures still exist. Workers could become exposed to LCs during other manufacturing process steps, such as during the formulation of the LC mixture (approximately 1.8 grams from Table 4-31), the handling and disposal of broken or defective panels, or during the recycling or disposal of an LCD at the end of its useful life. Because of the physical nature of the LCs (e.g., they are not volatile), typical worker exposures are likely to be through dermal or ingestion (e.g., accidentally ingesting LC present on a workers hands) routes. Worker exposure to LCs via dermal exposure is expected to be minimal for workers who wear gloves while handling LCs.

4.3.4.2 General population

Because of the enclosed nature of the LCD panel, it is unlikely that consumers could become exposed to LC compounds contained within the display through normal usage. LCs may be released into the environment should the panel become fractured, either through accident or through final disposal. No other releases of LC compounds into the environment were identified by the LCI data collected for the LCD, making it difficult to assess any possible exposures to nearby populations.

4.3.4.3 Ecological populations

Potential exposure to ecological populations could typically only occur through the migration of LCs through the environment after being released during manufacturing or disposal (either at LCD end-of-life, during disposal of broken or damaged panels during manufacturing, or during disposal of containers or equipment contaminated with LCs). The potential for transport of the LC through the environment is dependent on the identity of the chemical compound. The number of choices of LC compounds along with the unavailability of data on disposal quantities make it difficult to accurately assess the potential exposures which might result.

4.3.5 Human Health Effects

There are at least ten meaningful groups of liquid crystal compounds representing several hundred LC substances. Each of these substances is chemically unique, each having the potential to affect human health. Because of the number of LC compounds potentially present in the LCD, it is not possible to provide a comprehensive review of the human health effects of the universe of liquid crystal substances. However, a review of a small sample of LC compounds, those which appeared in the project LCI collected for the LCD, was conducted by EPA and the results presented below.

Toxicological testing of LC substances and mixtures was conducted by three manufacturers responsible for roughly 90% of LC production. Testing of their chemical products included testing for acute toxicity effects, and effects on skin and eyes. Results of the testing indicated that of 588 LC substances tested, only twenty-five LC compounds had an LD_{50}^{5} less

⁵ LD_{50} represents the dose of chemical which is lethal to 50% of the test population. By comparison, the LD_{50} for sodium chloride (table salt) is 3,000 mg/kg of body weight.

than 2,000 mg/kg of body weight (classified as exhibiting harmful effects to humans by the European Union), and only one had a LD_{50} of less than 200 (classified as toxic by the European Union). The remaining 562 LC substances did not have any acute toxic potential. Of the twenty-five harmful LCs, only twenty-two substances are produced and all are present at less than 10% concentration by weight, meaning that the resulting LC mixtures are not expected to exhibit harmful properties to humans (Becker, 2000). The remaining three harmful chemicals along with the lone toxic chemical were discontinued and excluded from further development. Several compounds, but still a minority, were found to be skin or eye irritants (Merck, 1999).

4.3.5.1 Chronic effects (noncancer)

EPA conducted a review of existing toxicological data for the LC substances listed in Table 4-31. The review failed to identify a RFD, RFC, NOAEL, or LOAEL for any of the LC substances shown. This typically indicates that insufficient testing of these chemical compounds has been performed to accurately determine their potential for chronic human effects.

4.3.5.2 Carcinogenicity and mutagenicity

An EPA review of toxicological studies for the liquid crystals identified in the life-cycle inventory for LCDs failed to identify an existing slope factor for any of the LC compounds. A lack of carcinogenicity data usually does not indicate that a compound is not carcinogenic, but only that sufficient testing to ascertain carcinogenicity has yet to be performed.

Toxicological testing of LC substances for mutagenic effects was conducted by three manufacturers responsible for roughly 90% of LC production. In a bacterial mutagenicity test of 615 LC substances, one LC compound showed mutagenic potential, with the remaining compounds displaying no mutagenic effects. The lone chemical that failed the test was excluded from further development and was never marketed (Becker, 2000). Additionally, 10 LC compounds representing each of the significant groups of LCs, underwent mutagenicity testing using mammalian cells. None of the tests indicated mutagenic activity. Based on both sets of data, the manufacturer concluded that there does not appear to be a suspicion of mutagenic potential in the liquid crystals it produces (Merck, 1999).

4.3.6 Environmental Regulations for Liquid Crystals

No regulations exist specifically for liquid crystals compounds. However, regulations may exist for individual liquid crystal compounds. Because of the number of possible LC compounds available for use in a LCD, a comprehensive review of U.S. regulations could not be provided.

4.4 CONCLUSIONS

The purpose of this chapter was to provide a more detailed analysis of a few select materials of interest to EPA and industry that was intended to better understand the potential exposures and chemical risks to both human and ecological populations. The materials selected for further analysis included lead, mercury, and liquid crystals, each selected for its known or suspected toxicity to humans and the environment, or because they are of particular interest to indsutry or the U.S. EPA. The analysis of each material summarized or evaluated the following key areas:

- Use of the materials in computer displays;
- Life-cycle inputs and outputs of the materials from computer displays;
- Life-cycle impacts associated with the material inputs and outputs;
- Potential exposures to the material including occupational, public, and ecological exposures;
- Potential human health effects;
- U.S. environmental regulations for the material; and
- Alternatives to reduce the use of the material in computer displays.

The following are the conclusions drawn from the analyses of lead, mercury, and liquid crystal use in the life cycle of both CRTs and LCDs.

4.4.1 Lead

Lead is found in glass components of CRTs, as well as in electronics components (printed wiring boards and their components) of both CRTs and LCDs. It is also a top priority toxic material at the U.S. EPA and the subject of electronics industry efforts to reduce or eliminate its use. The following conclusions were drawn from a focused look at lead's role in the life cycle of the computer display, and its effects on human health and the environment:

- Due to the much greater quantity of lead in the CRT than the LCD, lead-based life-cycle impacts from the CRT ranged from moderately to significantly greater than those from the LCD in every category, with the exception of solid waste landfill use. The most significant difference was in non-renewable resource consumption, where the CRT consumed over 40 thousand times the mass of non-renewable resources over the course of its life cycle than those consumed by the LCD. Other categories where CRTs had notably greater differences in impacts occurred in hazardous waste landfill use, chronic public health effects, and terrestrial toxicity.
- Contributions of lead-based impacts are small relative to the total life-cycle impacts from other materials in the CRT (e.g., glass, copper wire), with the greatest impacts from lead-based CRT outputs occurring in the categories of non-renewable resources, aquatic toxicity, and chronic public health effects (ranging from 0.1 to 0.2% of the overall impact scores in each category).
- For workers, inhalation is the most likely route of exposure to lead which may result in health concerns. General population exposure to lead is most likely to come from incidental ingestion of lead in the soil, or ingestion of lead brought into the household on

workers clothing or on shoes. Studies have discovered potentially high concentrations of lead in households within close proximity to certain facilities that use lead.

- Significant worker exposures to lead have been documented by existing studies of several processes which contribute to the life-cycle of the computer displays (e.g., lead smelting). These exposures have been as high as 90 times the OSHA recommended safety levels for exposure to workers at lead smelters. The resulting occupational chronic health effects to workers from lead exposure likely have been underestimated by the CDP LCIA methodology, which uses material inputs, and not outputs, as surrogates for exposure.
- Lead and lead compounds pose serious chronic health hazards to humans who may become over-exposed either in the workplace, or through the ambient environment. Lead exposure is associated with a range of adverse human health effects, including effects on the nervous system, reproductive and developmental problems, and cancer. Lead persists in the environment, but is relatively immobile in water under most surface and groundwater conditions.
- Alternatives are being developed, such as lead-free solders and glass components, that will potentially minimize the future lead content in both CRTs and LCDs.

4.4.2 Mercury

Mercury is contained within the fluorescent tubes that provide the source of light in the LCD. Mercury is also emitted from some fuel combustion processes, such as coal-fired electricity generation processes, which contribute to the life-cycle impacts of both CRTs and LCDs. EPA's concern with mercury and the potential for exposure during manufacturing and end-of-life processes warranted a more detailed analysis of mercury in the CDP. The following conclusions were drawn from a focused look at mercury's role in the life cycle of the computer display, and its effects on human health and the environment:

- The mercury emitted from the generation of power consumed by the CRT (7.75 mg) exceeds the entire amount of mercury emissions from the LCD, including both the mercury used in LCD backlights (3.99 mg) and the mercury emissions from electricity generation (3.22 mg). Although this was not expected because mercury is used intentionally in an LCD, but not in a CRT, the results are not surprising since mercury emissions from coal-fired power plants are known to be one of the largest anthropogenic sources of mercury in the United States. Because the CRT consumes significantly more electricity in the use stage than the LCD, its use stage emissions of mercury are proportionately higher than those of the LCD.
- Contributions from mercury-based impacts are not significant relative to the total lifecycle impacts from other materials (e.g., glass, copper wire) in the CRT or LCD, with the greatest impacts from mercury-based outputs occurring in the aquatic toxicity category (0.4% for CRTs, 0.01% for LCDs)
- Possible pathways of worker exposure during backlight fabrication include inhalation of mercury vapors, and dermal exposure or ingestion of mercury on skin. The most likely pathway for general population exposure is inhalation of mercury released into the air.
- Exposure data relevant to the manufacturing of mercury backlights were not available, therefore specific conclusions about the potential magnitude of worker exposures could not be made. Occupational chronic health effects to workers from mercury exposures

calculated during the impact assessment (3.99e-06 tox-kg for LCD, none for CRT) likely have been underestimated by the CDP LCIA methodology, which uses material inputs as surrogates for exposure.

- Mercury and mercury compounds pose serious chronic health hazards to humans who are exposed. EPA has determined that mercury chloride and methylmercury are possible human carcinogens. Mercury poses serious chronic health hazards to humans, affecting the nervous system, brain, and kidneys.
- Alternative backlights have been developed that not only eliminate mercury from the light, but also improve on many of the optical characteristics of the displays. Current development is focused on improving the energy efficiency of the alternative lights.

4.4.3 Liquid Crystals

Liquid crystals are organic compounds responsible for generating the image in an LCD. LCs are not present in CRTs. The toxicity of the LCs in LCDs has been alluded to in the literature, yet there is very little known about the toxicity of these materials. By including LCs in a more detailed analysis, this section attempted to better characterize any potential hazard and/or potential exposure of LCs from the manufacturing, use, and disposal of LCD monitors. The following conclusions were drawn from a focused look at LCs role in the life cycle of the computer display, and its effects on human health and the environment.

- LCs are combined into mixtures of as many as 20 or more compounds selected from hundreds of potential liquid crystal compounds. Because of the possible variations in mixtures and the sheer number of compounds available, a select number of liquid crystals were used to assess potential human health hazards.
- LCs do not appear to contribute significantly to any of the impact categories defined for this study. The total score for LCD occupational impacts based on potential worker exposure to LCs of 4.18 tox-grams, calculated using default toxicity values, represents less than 0.01% of the total overall chronic occupational health effects impact score of 898 tox-kg for the functional unit of one LCD.
- Impacts were not calculated for LC releases in the CDP LCIA because data regarding LC outputs were not available to the project. LCs are not used to fabricate CRTs and so have no environmental impacts in the CRT life cycle.
- Occupational exposures to LCs during the fabrication of the LCD panels are not expected to be significant. The enclosed nature of the chamber in which the LCDs are assembled, combined with the equipment (e.g., gloves, aprons) worn by workers in a clean room environment, are both expected to act to minimize exposures. Other occupational exposures may exist that have not been identified.
- Toxicological testing by a manufacturer of LC substances and mixtures showed that 95.6% (562 of 588) of the liquid crystals tested displayed no acute toxic potential to humans. Twenty-five of the remaining twenty-six chemicals had the potential to exhibit harmful effects to humans, while the remaining crystal was classified as toxic (EU classification) and thus was discontinued. An EPA review of toxicity data for the confidential LC compounds was unable to identify any relevant toxicity information. Insufficient toxicity data exist to assess the toxicity of specific LC compounds.

• Testing for mutagenic and carcinogenic effects by the supplier showed that 99.9% (614 out of 615) of the liquid crystal compounds tested displayed no mutagenic effects. The remaining chemical that showed mutagenic potential was excluded from further development. Additionally, mutagenicity testing of ten LC substances using mammalian cells showed no suspicion of mutagenic potential.

REFERENCES

- ACGIH (American Conference of Governmental Industrial Hygienists). 1991. *Documentation of the Threshold Limit Values and Biological Exposure Indices*. Sixth edition. ACGIH, Cincinnati, OH.
- ACGIH. 1998. Threshold Limit Values for Chemical Substances and Physical Agents Biological Exposure Indices. ACGIH, Cincinnati, OH.
- ATSDR (Agency for Toxic Substances and Disease Registry). 1999. Toxicological Profile for Mercury. U.S. Department of Health and Human Services. Web site available at http://www.atsdr.cdc.gov/toxprofiles/tp46.html. March.
- ATSDR. 1999. Toxicological Profile for Lead. U.S. Department of Health and Human Services. Web site available at: http://www.atsdr.cdc.gov/toxprofiles/tp46.html. March
- Barnes, D.G., and M. Dourson. 1988. "Reference Dose (RfD): Descriptions and Uses in Health Risk Assessments." *Regulatory Toxicology and Pharmacology*. **8**: 471-486.
- Bress, W.C., and Bidanset, J.H. 1991. "Percutaneous *in vivo* and *in vitro* absorption of lead." *Veterinary and Human Toxicology.* **33**: 212-214.
- Becker, W. 2000. "Results of the Toxicological Testing of Liquid Crystal Compounds." Presented at International Display Works Conference, Kobe, Japan. December.
- Becker, W. 2001. Personal communication by e-mail with W. Becker, Merck, Liquid Crystals Division, and M. Socolof, University of Tennessee, Center for Clean Products and Clean Technologies. December 6.
- Busio, M. and Steigelmann, O. 2000. "New Frit Glasses for Displays." *Glasstech*. Berlin Glass Science and Technology. **73**(10): 319-325.
- Doggett, T. 2001. "EPA Seeks to Dismiss Utility Lawsuit over Mercury." Reuters press release dated April 12, 2001 accessed at http://www.planetark.org/dailynewsstory.cfm?newsid=10496&newsdate=12-Apr-2001. September 2.
- Donofrio, R. and Eckel, R. 1999. "Arc Lamps for Microelectronics." Presented at Vehicle Displays and Microsensors '99 SID Symposium, pp.103-111.
- EIAJ (Electronic Industries Association of Japan). 1996. 1996 Update to: Research Report on the Visions of the Electronic Display Industry to the Year 2000. Translated and published in English by InterLingus.

- EPA (Environmental Protection Agency). 1986. EPA Air Quality Criteria for Lead. Publication No. EPA 600/8-83-028af. U.S. EPA, Office of Research and Development, Office of Health and Environmental Assessment. Research Triangle Park, NC.
- EPA. 1998. Fact Sheet: "Utility Air Toxics Report to Congress." U.S. EPA, Office of Air Quality, Planning, and Standards. Web site available at: http://www.epa.gov/ttn/oarpg/utilfs.html. February 25.
- EPA. 1999. "Proposed Rule Change; Lead and Lead Compounds, Lowering of Reporting Thresholds, Community Right-to-Know Toxic Chemical Release Reporting." Federal Register: Volume 64, Number 148. U.S. EPA, Office of Pollution Prevention and Toxics, Washington, D.C. August.
- EPA. 2000a. "Mercury Study Report to Congress: White Paper." U.S. EPA Office of Air and Radiation. Web site available at http://www.epa.gov/oar/merwhite.html. (The full eight-volume 1997 *Mercury Report to Congress* is available online at: http://www.epa.gov/airprogm/oar/mercury.html). September 26
- EPA. 2000b. "EPA Decides Mercury Emissions from Power Plants Must be Reduced." U.S. EPA Headquarters Press Release. December 14.
- EPA. 2001a. "Mercury Study Report to Congress: Overview." EPA Office of Air and Radiation. Web site available at: http://www.epa.gov/oar/mercover.html. (The full eightvolume 1997 *Mercury Report to Congress* is available online at: http://www.epa.gov/airprogm/oar/mercury.html). February 8.
- EPA. 2001b. "Persistent and Bioaccumulative Toxic Chemical Program's Chemical List." U.S. EPA, Office of Pollution Prevention and Toxics. Web site available at: http://www.epa.gov/opptintr/pbt/cheminfo.htm. March 8.
- EPA. 2001c. "National Advisory on Mercury in Fish Caught by Family and Friends: for Women Who Are Pregnant or May Become Pregnant, Nursing Mothers, and Young Children." U.S. EPA, Office of Water, EPA-823-F-01-004. January, 2001.
- FDA (U.S. Food and Drug Administration.). 2001. "Consumer Advisory: an Important Message for Pregnant Women and Women of Childbearing Age Who May Become Pregnant about the Risks of Mercury in Fish." U.S. Food and Drug Administration, Center for Food Safety and Applied Nutrition. March.
- Gosselin, R.E., R.P. Smith, and H.C. Hodge. 1984. *Clinical Toxicology of Commercial Products*. 5th ed. Williams and Wilkins, Baltimore, MD.
- HSDB (Hazardous Substance Data Base). 2001. MEDLARS Online Information Retrieval System. National Library of Medicine, National Toxicology Program (via TOXNET). Bethesda, MD.

- IARC (International Agency for Research on Cancer). 1987. IARC Monographs on the Evaluation of the Carcinogenic Risk of Chemicals to Humans: Overall Evaluations of Carcinogenicity. Supplement 7. World Health Organization, Lyon, France.
- IRIS (Integrated Risk Management System). 1999. U.S. Environmental Protection Agency. National Center of Environmental Assessment. Office of Research and Development. Cincinnati, OH.
- Keenan, C. and Kellett, T. 2001. Personal communication with C. Keenan and T. Kellett, Abt Associates Incorporated, and K. Hart, U.S. EPA. Background Research on Lead-Free Solder Alternatives.
- Lee, C., Chang, S., Wang, K., and Wen, L. 2000. "Management of Scrap Computer Recycling in Taiwan." *Journal of Hazardous Materials*. **A73**: 209-220.
- Menad, N. 1999. "Cathode Ray Tube Recycling." *Resources, Conservation, and Recycling.*26: 143-154.
- MERCK. 1999. "Toxicological Investigations of Liquid Crystals." Presented at 28th Freiburger Arbeitstagung. March 24-26.
- Monchamp, A., Evans, H., Nardone, J., Wood, S., Proch, E., Wagner, T. 2001. "Cathode Ray Tube Manufacturing and Recycling: Analysis of Industry Survey." *Electronic Industries Alliance*. Spring.
- Musson, S., Jang, Y., Townsend, T., and Chung, I.. 2000. "Characterization of Lead Leachability from Cathode Ray Tubes Using the Toxicity Characteristic Leachate Procedure." *Environmental Science and Technology.* 34: 4376-4381.
- NIOSH (National Institute for Occupational Safety and Health). 1997. *NIOSH Pocket Guide to Chemical Hazards*. Publication No. 97-140. U.S. Government Printing Office. Washington D.C.
- OSRAM. 1998. "The World's First Mercury-Free Flat Lamp." Web site available at: http://www.osram.com/txt/aktuell/planone.htm. July 13.
- Sittig, M. 1985. *Handbook of Toxic and Hazardous Chemicals and Carcinogens*. 2nd ed. Noyes Data Corporation, Park Ridge, NJ.
- Srivastava, A.M. and Sommerer, T.J. 1998. "Fluorescent Lamp Phosphers." *Electrochemical Society Interface.* Summer.
- Techneglas, Incorporated. "Frit Facts: A Brief Technological Summary of Televison Solder Glass for CRT Technicians, Engineers, and Managers", downloadable fact sheet. http://www.techneglas.com.

Chapter 5

SUMMARY AND CONCLUSIONS

The purpose of the CDP, as stated in Chapter 1, is to provide a scientific baseline of lifecycle environmental impacts of CRTs and LCDs, help manufacturers identify areas to focus improvement assessment activities, and to develop a life-cycle model for future analyses. The primary targeted audience is the electronics industry, for whom results may provide insight into improvement opportunities in the life cycle of CRTs and/or LCDs. In addition, the general public may also find results useful when considering environmental impacts of each display type. This chapter briefly summarizes the results and draws conclusions based on those results. This report, however, does not include direct comparative assertions or improvement assessments based on the results. Alternatively, results and conclusions are described in terms of the overall LCI versus the LCIA, and details of the impact assessment, including the additional assessments of lead, mercury and liquid crystals, and the sensitivity analyses. Major uncertainties, cost and performance considerations, suggestions for improvement opportunities, and suggestions for further research are also provided.

5.1 LCI vs. LCIA

In this LCA, a life-cycle inventory (LCI) was compiled from many data sources, including both primary and secondary data sources. The primary data were obtained from component and monitor manufacturers of CRTs and LCDs. In an LCA, inventory data provide information on how much material is being consumed in the life cycle (i.e., inputs) and how much material is generated/released (i.e., outputs). The LCI results of this report are detailed in Chapter 2. The LCI provides inventory data grouped by inventory type (e.g., primary material, energy, air emission, solid waste).

The LCI alone, however, does not always translate directly into impact categories that may be of interest. That is, a given amount of one material may have different impacts (for a certain impact category) than the same amount of another material. Furthermore, some materials may affect more than one impact category. For example, an air emission could affect air acidification as well as being toxic to humans breathing it. Therefore, a life-cycle impact assessment (LCIA) is conducted to reveal potential impacts in several impact categories. In this CDP LCIA, described in detail in Chapter 3, impacts are sometimes driven by materials other than the top inventory contributors. For example, the top air emission for LCDs is carbon dioxide (Table 2-49), however the greatest global warming impact score is from SF_6 in the LCD monitor/module manufacturing process (Table 3-25).

To illustrate that the inventory results may not directly translate into impact results, the first two columns in Table 5-1 show which monitor has greater inventory amounts for each inventory type in the LCI, and the last two columns show which monitor has greater impact scores for each impact category in the LCIA. The impact categories that are affected by each inventory type are in the same rows as the associated inventory type. As seen in Table 5-1, some impact categories associated with ancillary material and water pollutant inventory types had opposing outcomes in the LCI versus the LCIA. For example, the three impact categories affected by the ancillary material inventory had greater impacts for the CRT, although the

ancillary material inventory had greater amounts of inputs for the LCD. In this case, both primary and ancillary materials contribute to the impact categories, causing differing results.

Considering the wastewater outputs, which are greater for the LCD than the CRT, the impacts related to water releases are in some cases greater for the CRT than the LCD. Note that although the wastewater volume is greater for the LCD, the total mass of water pollutants in the LCI is greater for the CRT (see Table 2-24). In the LCIA, the LCD has greater impacts for water eutrophication and aquatic toxicity, but not for the two water quality categories (BOD and TSS), chronic health effects to the public, nor terrestrial toxicity, all of which include water emissions in calculating the impact score.

Baseline life-cycle inventory (LCI)		Baseline life-cycle impact assessment (LCIA)		
Inventory type	Monitor with greater inventory results	Potential impact category(ies) associated with inventory type	Monitor with greater impact results	
Primary materials	CRT	Renewable resource use	CRT	
		Nonrenewable resource use	CRT	
		Chronic health effects, occupational	CRT	
Ancillary materials	LCD	Renewable resource use	CRT	
		Nonrenewable resource use	CRT	
		Chronic health effects, occupational	CRT	
Water inputs	CRT	Renewable resource use	CRT	
Fuel inputs	CRT	Energy use	CRT	
		Chronic health effects, occupational	CRT	
Electricity inputs	CRT	Energy use	CRT	
Total energy inputs	CRT	Energy use	CRT	
Air pollutant outputs	CRT	Global warming	CRT	
		Ozone depletion	а	
		Photochemical smog	CRT	
		Acidification	CRT	
		Air particulates	CRT	
		Chronic health effects, public	CRT	
		Aesthetics (odor)	CRT	
	Í	Terrestrial toxicity	CRT	

Table 5-1	Baseline LCL	vs LCIA.	monitor with	orester inventory	y amount and impact
1 abic 3-1.	Daschill LUI	VS. LUIA.	IIIUIIIIUI WIUI 2		γ amount and impact

Baseline life-cycle inventory (LCI)		Baseline life-cycle impact assessment (LCIA)			
Inventory type	Monitor with greater inventory results	Potential impact category(ies) associated with inventory type	Monitor with greater impact results		
Wastewater outputs	LCD	none	NA		
Water pollutant outputs	CRT	Water eutrophication	LCD		
		Water quality, BOD	CRT		
		Water quality, TSS	CRT		
		Chronic health effects, public	CRT		
		Aquatic toxicity	LCD		
		Terrestrial toxicity	CRT		
Hazardous waste outputs	CRT	Hazardous waste landfill use	CRT		
Solid waste outputs	CRT	Solid waste landfill use	CRT		
Radioactive waste outputs	CRT	Radioactive waste landfill use	CRT		
Radioactivity outputs	CRT	Radioactivity	CRT		

Table 5-1.	Baseline	LCI vs.	LCIA:	monitor	with	greater	inventory	amount	and im	pact
------------	----------	---------	-------	---------	------	---------	-----------	--------	--------	------

^{*a*} The LCIs for both the CRT and LCD contain data for substances that were phased out of production by 1996 due to their ozone depletion potential. Whether these emissions still occur in countries that were signatories to the Montreal Protocol and its Amendments and Adjustments (such as the United States and Japan) is not known, but considered to be unlikely. When phased-out substances are included in the inventory, the CRT has greater ozone depletion impacts than the LCD. However, if phased-out substances are removed from the inventories, the results are switched, with the LCD having greater impacts.

5.2 LCIA RESULTS

5.2.1 CRT and LCD Baseline Results

The LCIA results, presented in detail in Chapter 3, showed that the CRT has greater total life-cycle impact indicators in most of the impact categories (see Table 3-10). In the baseline scenario, the CRT has greater impacts than the LCD in all but two impact categories (eutrophication and aquatic toxicity). However, note that for the ozone depletion category, the LCIs for both the CRT and LCD contain data for substances that were phased out of production by 1996 due to their ozone depletion potential. Whether these emissions still occur in countries that were signatories to the Montreal Protocol and its Amendments and Adjustments (such as the United States and Japan) is not known, but considered to be unlikely. When phased-out substances are included in the inventory, the CRT has greater ozone depletion impacts than the LCD. However, if phased-out substances are removed from the inventories, the results are switched, with the LCD having greater impacts.

When considering which life-cycle stage has greater impacts, the LCIA results showed that the manufacturing life-cycle stage dominates impacts for most impact categories for both the CRT and LCD (refer to Section 3.3). Table 5-2 summarizes which life-cycle stages have the greatest impacts for each impact category for the CRT and LCD. As shown in Table 5-2, the CRT has nine and the LCD has 11 impact categories with greatest impacts from the manufacturing life-cycle stage. Only six categories (solid waste landfill use, global warming, ozone depletion, acidification, air particulates, and chronic public health) have greatest impacts from the CRT use stage, and four categories (solid waste landfill use, acidification, air particulates, and chronic public health) have greatest impacts from the LCD use stage. The CRT has three categories with greatest impacts from the upstream life-cycle stage and the LCD has three. The end-of-life (EOL) life-cycle stage is greatest for the same two impact categories for both the CRT and LCD (hazardous waste landfill use and radioactive waste landfill use). Note that the EOL stage impacts are generally very small contributors to the overall impacts. This is likely because of the small inventories associated with the EOL processes, but also may be a function of the incomplete and/or secondary data for the EOL (i.e., no remanufacturing data, and secondary data not completely specific to the monitors evaluated in this study).

A more detailed evaluation of lead, mercury, and liquid crystals was completed in Chapter 4. As expected, the CRT, which has lead in the glass, frit, and printed wiring boards (PWBs), has greater impacts from lead than did the LCD, which only has lead in the PWBs. Regarding mercury, there were greater inventories of mercury in the CRT life cycle than in the LCD life cycle, despite the fact that only the LCD has mercury directly in the product. The greater amount of mercury is from the release of mercury and mercury compounds from the generation of electricity. And as the CRT life cycle uses more electricity than the LCD, there was a greater quantity of mercury releases reported for the CRT than the LCD. Liquid crystals are only found in LCDs, and therefore, there are no associated impacts for the CRT. Little conclusive information was available on the liquid crystal materials. A detailed literature search was conducted, however very little data were available on the toxicity of these materials. Based on the limited toxicity data obtained, liquid crystals currently do not appear to be a significant human health or environmental hazard in the LCD life cycle. However, there were insufficient toxicity data available to make a definitive conclusion about liquid crystal toxicity.

Impact category	Monitor type with greatest impacts				
	Upstream	Manufacturing	Use	EOL	
Renewable resource use	LCD	CRT			
Nonrenewable resource use		CRT, LCD			
Energy use		CRT, LCD			
Solid waste landfill use			CRT, LCD		
Hazardous waste landfill use				CRT, LCD	
Radioactive waste landfill use				CRT, LCD	
Global warming		LCD	CRT		
Ozone depletion		LCD	CRT		
Photochemical smog	LCD	CRT			
Acidification			CRT, LCD		
Air particulates			CRT, LCD		
Water eutrophication	CRT	LCD			
Water quality, BOD		CRT, LCD			
Water quality, TSS		CRT, LCD			
Radioactivity	CRT, LCD				
Chronic health effects, occupational		CRT, LCD			
Chronic health effects, public			CRT, LCD		
Aesthetics (odor)		CRT, LCD			
Aquatic toxicity	CRT	LCD			
Terrestrial toxicity		CRT, LCD			
TOTALS	CRT=3 LCD=3	CRT=9 LCD=11	CRT=6 LCD=4	CRT=2 LCD=2	

 Table 5-2. Monitor type with greatest impacts for each life-cycle stage and impact category (baseline scenario)

5.2.2 CRT Results

For the CRT, many of the impacts were driven by a single material in the inventory. As stated in Section 3.3.15 and shown in Table 3-57, in 14 of the 20 impact categories, the top individual contributor to the impacts was responsible for greater than 50% of the impacts. This shows that the CRT data are highly sensitive to a few data points. Major conclusions from the CRT LCIA are as follows:

• Energy used in glass manufacturing and associated production of LPG are driving the baseline CRT results (they dominate ten impact categories, including overall life-cycle energy use).

5.2 LCIA RESULTS

- The large amounts of fuel used as energy sources are driving occupational health effects. Occupational impacts are calculated from inventory input amounts, and therefore there may or may not actually be exposure to these fuels (e.g., they may be contained); however, the results illustrate the potential for health effects, especially under spill or upset conditions.
- The generation of electricity for the use stage dominates seven impact categories.
- Air emissions of sulfur dioxide from electricity generation (for the use life-cycle stage) drive chronic public health effects, acidification, and terrestrial toxicity impacts. This may be a concern, for example, in areas in nonattainment of regulated levels of sulfur dioxide in the United States.

The use of LPG fuel in glass manufacturing dominated ten impact categories: two directly from the LPG used in glass/frit manufacturing (energy use impacts and chronic occupational health effects) and eight from LPG production (renewable resource use, nonrenewable resource use, photochemical smog, air particulates, water eutrophication, BOD water quality, TSS water quality, and aesthetics). In addition, impacts from the generation of electricity during the use stage dominated seven impact categories: solid waste landfill use, radioactive waste landfill use, global warming, ozone depletion, acidification, chronic public health, and terrestrial toxicity. The CRT tube manufacturing process, which represents the most functionally and physically (by mass) significant component of the CRT monitor, only dominated one impact category (aquatic toxicity). Twenty-six percent of the aquatic toxicity score was from phosphorus outputs from tube manufacturing, while most of the rest were from the materials processing life-cycle stage. The remaining two impact categories (hazardous waste landfill use and radioactivity) had greatest impacts from the landfilling of the assumed hazardous proportion of CRT monitors, and the release of Plutonium-241 in steel production, respectively (Table 3-57). The radioactivity impacts are driven by the radionuclide Pu-241, due to the electric grid inventory included in the steel production secondary data set, which includes nuclear fuel reprocessing.

The large amount of LPG reported for glass manufacturing was originally questioned during the data collection and verification stage of this project. While no compelling reason could justify removing the LPG data in the baseline case, a sensitivity analysis was conducted in which the glass energy data were modified. Other sensitivity analyses were also conducted (i.e., manufactured life, modified LCD monitor manufacturing energy, and modified LCD EOL distributions). However, the only scenario that substantially altered the comparative results was the modified glass energy scenario (see Table 3-62 and Section 3.4.5).

The overall energy in the baseline scenario was nearly seven times greater than that in the modified scenario (from 20,800 MJ to 3,020 MJ), and the amount of LPG dropped to zero, while other energy sources increased. The basis for the modified data was removing the energy inputs from one suspect data set. As a result, the CRT modified glass energy scenario had greater energy use impacts in the *use* stage than in the *manufacturing* stage for the CRT. The amount of LPG used in glass manufacturing in the baseline scenario is 351 kg/monitor of LPG, which alone

costs about \$71.¹ This is a significant amount of the cost of a complete CRT monitor (the range of a few currently selling 17" CRTs is \$158-316, and the average cost from primary data collected in the CDP was \$541, which are presented in Section 5.4). Therefore, it is likely that the actual energy inputs to the glass manufacturing process is somewhere between the baseline and modified glass scenarios. In conclusion, more information is needed on energy used in glass manufacturing, which is driving CRT baseline results.

The additional analyses for the CRT of lead and mercury also revealed that the use of lead could present health risks, but the method of using only inputs to evaluate occupational impacts (see Section 3.1.2.13) may not adequately represent occupational exposures and risks. Further refinement of the occupational impact analysis may be warranted.

Although there is no mercury in the CRT monitor, mercury emissions from electricity generation in the CRT life cycle were greater (in mass) than the mercury used in the LCD. Therefore, to reduce mercury emissions from the CRT life cycle, efforts to reduce electricity consumption could be taken. Additionally, changes to the electric grid could also reduce mercury emissions from the CRT life-cycle.

5.2.3 LCD Results

The LCD impact results were less sensitive to an individual input or output than the CRT results, although in 11 of the 20 impact categories an individual input was still responsible for greater than 50% of the total impacts (Table 3-58). In general, the LCD results are less uncertain than the CRT results. This is because most of the CRT results are being driven by either glass input data or data from secondary sources, while LCD impacts are being driven more by data from primary sources. Some results to note are as follows:

- The LCD monitor/module manufacturing process group had greatest impacts in six impact categories (Table 3-58).
- Although the top contributor to the energy impact category was electricity consumed in the use stage (30%), the overall energy impacts were greater from the manufacturing stage than the use stage.
- In the glass energy sensitivity scenario, the use stage had greatest energy impacts, although only by a small margin over the manufacturing stage (see Figure 3-26).
- Sulfur dioxide [emitted from electricity generation for the use stage, and constituting only 0.37% of the air emission inventory (see Table 2-49)] dominates the acidification, chronic public health, and terrestrial toxicity impact categories (Table 3-58). The high public health and terrestrial toxicity scores are due to its low non-cancer toxicity value and resulting high hazard value (HV).
- Sulfur hexafluoride (SF₆) from LCD monitor/module manufacturing was the single greatest contributor to the global warming impact score; however, carbon dioxide from the use stage and the materials processing stage also contributed significantly to the global warming impacts (Table 3-25).

¹ Based on a "daily market price" on August 29, 2001, of \$0.4160/gallon of LPG (http://www.americanpowernet.com/pub_energy/futures.html). For 351 kg/functional unit in the CRT manufacturing life-cycle stage, the cost is about \$71 per functional unit (i.e., one monitor), assuming a density of LPG of 2.053 kg/gallon. For the LCD, the 16.8 kg/functional unit of LPG would cost about \$3.40 per monitor.

5.2 LCIA RESULTS

- The glass energy inputs did not directly dominate any impact categories, as they did for the CRT (due to the smaller mass of glass in the LCD); however, LPG production (required for the glass energy fuel) dominated two categories: TSS water quality and aesthetics (Table 3-58).
- LNG as an ancillary inventory material was questionably very large and had greatest impacts in two categories: nonrenewable resource use and photochemical smog (Table 3-58; shown there as "Natural Gas Production" due to that process being used as a surrogate for LNG production).

The additional analyses of lead, mercury, and liquid crystals showed that the LCIA alone is not adequate enough to determine all the potential impacts within the life-cycle of the LCD monitors. Similar to the conclusion for the CRT, lead-based occupational impacts would require further refinement of the LCIA methodology. The LCIA method in this LCA used inputs as surrogates for occupational exposure. There are outputs, within the occupational setting, that should also be considered.

For mercury, which is found in the backlights of the LCD monitors, there is nearly the same amount of mercury by mass emitted to the air during electricity generation as there is mercury used to make the backlight unit. The mass of mercury input for backlights is only about 20% greater than the mercury air emissions from electricity generation (across all life-cycle stages).

Liquid crystals were also identified by the CDP Core Group as a material for which additional information would be reviewed. The LCIA did not find the liquid crystals to be significant contributors to any impact categories; however, this could partially be due to the lack of information on them. The additional analysis also revealed limited information, but qualitatively, did not show significant potential risk.

5.2.4 CRT vs. LCD Sensitivity Analysis Results

The only sensitivity analysis to show significant difference in the results was the modified glass energy scenario. In comparing the CRT and LCD, the CRT *baseline* scenario had greater impacts than the LCD in all but two impact categories (eutrophication and aquatic toxicity) and possibly three (ozone depletion). In the *modified glass energy scenario*, nine of the 20 categories had greater impacts from the LCD life-cycle than the CRT. Energy use remained greater for the CRT; however, nonrenewable resource use, global warming, photochemical smog, eutrophication, BOD and TSS water quality, chronic occupational health effects, and aesthetics all reversed such that the LCD had greater impacts than the CRT (Table 3-62). As stated above, it is believed that a more true representation of the monitor life cycles lies somewhere between the baseline and modified glass energy scenario. Further work is recommended in clarifying and refining glass energy input information.
5.3 UNCERTAINTIES

As with any LCA, it is not uncommon for there to be uncertainty associated with such a large data collection effort. Two of the largest sources of uncertainty in this LCA that have a significant effect on the results are as follows:

- *CRT and LCD glass manufacturing energy inputs (from primary data)*: The larger amount of glass used in CRTs than LCDs results in the CRT having greater associated uncertainty than the LCD results.
- Secondary data for upstream and fuel production processes: When any one material is used in the life-cycle of either monitor in large quantities, the impacts associated with the inputs and outputs from the production of that material may become significant. For example, LPG and LNG production were both used in significant enough amounts to influence some impact categories. Therefore, the uncertainty in the secondary data becomes important. This highlights the need for a consistent, national (or international) LCI database that is updated regularly.

Other uncertainties associated with individual data points collected from primary data sources may be found in the data for this analysis. However, they had less effect on the overall results than the uncertainties mentioned above. For manufacturers interested in conducting improvement assessments, closer review of such uncertainties may be warranted.

Other uncertainties in the LCA pertain to uncertainties inherent in LCIA methodology. The purpose of an LCIA is to evaluate the *relative potential* impacts of a product system for various impact categories. There is no intent to measure the *actual* impacts or provide spatial or temporal relationships linking the inventory to specific impacts. Uncertainties are inherent in each impact category, and the reader is referred to the baseline LCIA results in Section 3.3 for a detailed discussion of uncertainties by impact category.

Another point that should be recognized in the overall comparison of CRTs and LCDs is that CRTs are a more mature technology than LCDs. Changes in LCD manufacturing processes have likely occurred during the development and publication of this report. Therefore, comparisons must be carefully drawn when evaluating the mature CRT to the newer LCD technology.

5.4 COST AND PERFORMANCE CONSIDERATIONS

The focus of this study has been on the environmental effects associated with CRTs and LCDs. The environmental attributes or burdens of a product are not expected to be considered alone when evaluating the marketability and commercial success of a product. The cost and performance of each monitor type are obviously critical components to a company's or consumer's decisions of whether to produce or purchase a product. This section briefly addresses a few direct costs associated with the monitors. A complete cost analysis, including all direct costs (e.g., material costs) and indirect costs (e.g., environmental costs to society) are beyond the scope of this report. Direct retail costs of the monitors and electricity costs are presented herein.

The average retail price of 1997-2000 model year monitors, collected from the manufacturers who supplied data for this project, as well as the performance information, are presented in Table 5-3. Costs collected from current monitors on the market are presented in Table 5-4. From Table 5-3, which represents primary data collected on the actual monitors included in this study, the LCD is approximately 2.7 times more costly. More recent data show that prices have come down, and the difference in prices between the CRT and LCD has also been reduced.

Monitor	Display Size	Resolution	Brightness range	Contrast ratio range	Number of Colors	Average cost from primary data
	(inches)	(pixels)	(cd/m^2)			(US\$)
CRT monitor (functional unit aggregate)	17	1024x768	86-154		"Full color"	\$541
LCD monitor (functional unit aggregate)	15	1024x768	200-300	200:1 - 300:1	"Full color"	\$1,450

Table 5-3. Primary cost and performance data collected from manufacturers for the CDP

---- Not reported or not applicable.

A complete cost analysis would require assessing the costs from each life-cycle stage. The costs presented above are retail costs that presumably represent the manufacturing costs, but probably not external environmental costs, for example. The costs from the use stage can be represented by the electricity costs during the use stage. The average cost of residential and commercial electricity in the United States is approximately \$0.021/MJ,² and the CRT and LCD monitors use about 2,290 and 853 MJ/functional unit, respectively, in the use stage baseline scenario, which assumes a total of 13,547 hours per life over a period of 6.5 years (see Section 2.4.1.2). Therefore, the electricity costs to consumers during the use stages are \$48 for the CRT and \$18 for the LCD. The amount of electricity consumed and the associated cost of that electricity for each life-cycle stage in the baseline scenario are presented in Table 5-5.

² This number was calculated from a value found at the following Web address: www.eia.doe.gov/cneaf/electricity/ esr/t11.txt.

Monitor	Display Size	Resolution	Brightness	Contrast ratio	Number of	2001 Cost
	(inches)	(pixels)	(cd/m ²)		Colors	(\$US)
CRTs	• • •	• • •	•		•	•
Monitor 1	17/16	1280x1040				\$158
Monitor 2	17/16.1	1280x1040		"High contrast, anti-static, anti-glare coating."		\$171
Monitor 3	17/16	1600x1200				\$316
LCDs						
Monitor 1	15.1	1024x768		200:1	16.7 million	\$349
Monitor 2 ^{<i>b</i>}	15.1	1024x768	200	250:1	16.7 million	\$400
Monitor 3	15.1	1024x768	200 ^c	200:1 ^c	16+ million	\$439
Monitor 4 ^b	15.1	1024x768	200	250:1	16.7 million	\$499
Monitor 5	15	1024x768	210 ^c	350:1		\$554

Table 5-4. Cost and performance data for some currently selling CRTs and LCDs^a

^{*a*} All information from Vol. EC23 of the eCOST.com catalog, except where noted otherwise.

^b Data from the manufacturer's Web site except for prices, which where obtained from http://www.cdw.com on 8/29/01.

^{*c*} Data are from the manufacturer's Web site.

---- Not reported or not applicable.

	CRT			Ι	LCD	
Life-cycle stage	Electricity use (MJ/functional unit) (see Table J-3)	unit cost (\$/MJ)	Cost (\$US)	Electricity use (MJ/functional unit) (see Table J-12)	unit cost (\$/MJ)	Cost (\$US)
Upstream	73.2	0.012 ^a	\$1.3	8.55	0.012 ^a	\$0.10
Manufacturing	129	0.012 ^a	\$1.5	278	0.012 ^a	\$3.4
Use	2,290	0.021 ^b	\$48	853	0.021 ^b	\$18
EOL	0.229	0.012 ^a	\$0.003	0	0.012 ^a	0
Total	2,492		\$51	1,140		\$22

Table 5-5. Life-cycle electricity costs (baseline scenario)

^{*a*} 1999 U.S. average cost of electricity for the industrial sector is 0.0443/kWh. Assuming 3.6 MJ/kWh, (0.0443/kWh)/(3.6 MJ/kWh) = 0.012/MJ. Source: www.eia.doe.gov/cneaf/electricity/esr/t11.txt. Note that the use of the *U.S.* average cost is simply to compare costs among life-cycle stages, although the actual costs would be mostly from *Asia*.

^b 1999 U.S. average cost of electricity for the residential and commercial sectors is \$0.0771/kWh. Assuming 3.6 MJ/kWh, (\$0.0771/kWh)/(3.6 MJ/kWh) = \$0.021/MJ. Source: www.eia.doe.gov/cneaf/electricity/esr/t11.txt.

5.4 COST AND PERFORMANCE CONSIDERATIONS

The LCA is defined such that the monitor assessments are performed on a functionally equivalent basis. To the extent possible, data were collected on functionally equivalent monitors. The data presented in Table 5-3 are the range of specifications provided by the monitor assemblers, but not necessarily from the manufacturers of all of the component parts. Therefore, we are unsure if the specifications provided by the monitor assemblers also represent those of the component parts, since the component parts manufacturers did not consistently supply performance data as requested in the data collection questionnaire used in this study. However, when companies were approached to participate in the study, they were informed of the performance specification parameters within which the study boundaries were defined. Therefore, it is assumed that they meet the specifications as presented in Chapter 1 (Table 1-2) and in Table 5-3 and they perform relatively equivalently. In the primary data, the reported brightness of the CRT was less than the LCD, otherwise, they are functionally similar.

5.5 IMPROVEMENT ASSESSMENT OPPORTUNITIES AND TARGETED AUDIENCE USES OF REPORT

To meet the primary objective of providing the display industry with data to perform improvement assessments, the industry should look at the manufacturing life-cycle stage, while recognizing the influences of the other stages. CRT improvement opportunities could include improved energy efficiency during glass manufacturing and display use, as well as reductions in lead content. LCD improvement opportunities could also include improved energy efficiency, especially during manufacturing. Certain materials, such as SF_6 and its contribution to global warming, may also be of concern and an area to focus on in future improvement assessments.

In addition, any improvement assessment should consider how changes in one life-cycle stage will affect impacts in other stages. For example, in Chapter 4 we saw that the mercury inputs and outputs from the intentional use of mercury in an LCD backlight are less than (by mass) the mercury emissions from the CRT use stage, due to the relatively high energy usage by the CRT and the emissions of mercury from electricity generation. In this example, we can see that on a pure mass basis, a product's energy efficiency is a key consideration and any changes in manufacturing should consider if it will affect changes in use. However, this project did not conduct a quantitative risk assessment to determine where the greatest potential for mercury exposure (and therefore risk) might occur. Consequently, we cannot definitively say whether it is better to have a potentially less energy-efficient backlight that does not contain mercury or a more energy-efficient backlight that does. Nonetheless, this analysis highlights the life-cycle trade-offs that must be considered in an improvement assessment.

Another objective of this study was to provide an LCA model for future analyses. Companies or individuals who have more current data for the CRT or LCD can apply them to the model presented here. For example, changes in an individual process can be identified and incorporated into the model. The other processes that are not expected to change significantly can be left unchanged, and only limited data would need to be altered. This would reduce the time and resources that would normally be required for a complete analysis.

Finally, those interested in comparing the results of the two monitors can apply their own set of importance weights to each impact category to determine their individual decision. For example, if energy impacts are much more important than aesthetics to a particular person, they can weigh energy more heavily in concluding which monitor may have fewer environmental impacts, while keeping in mind the data limitations and uncertainties, as well as cost and performance considerations.

5.6 SUGGESTIONS FOR FUTURE RESEARCH

5.6 SUGGESTIONS FOR FUTURE RESEARCH

Areas where future research could be conducted to refine and/or continue the use of the results in this study are as follows:

- gather more information on energy use in glass manufacturing;
- develop consistent materials and fuel processing data in a national (or international) LCI database that is updated regularly;
- refine and/or update some of the LCD manufacturing data (e.g., LNG data);
- collect more complete EOL data (e.g., remanufacturing data, and primary data for incineration and landfilling) to determine better representation of the EOL impacts;
- conduct more research on EOL options for LCDs;
- collect more detailed data on landfilling and other treatment processes, such as water treatment where no impacts were calculated;
- update manufacturing data to meet more recent monitor model years;
- conduct a more focused analysis on selected areas for detailed improvement assessments; and
- evaluate process changes or other alternatives against an "average 1997-2000 model year" to evaluate impacts of changes or improvements over time.

APPENDIX A

FLAT PANEL DISPLAY TECHNOLOGIES

this page intentionally left blank

FLAT PANEL DISPLAY TECHNOLOGIES

1. Background

Flat panel displays (FPDs) are increasingly gaining a presence in the computer display market. They provide, for example, a more compact display as used in laptop computers and are viable substitutes for cathode ray tube (CRT) displays. Other advantages over the CRT are higher contrast, sunlight readable, more reliable, and more durable (i.e., require much less maintenance) Koch and Keoleian, 1995). In general, the major disadvantages have been that the resolution and quality of the image did not match that of CRTs. Several different types of FPD technologies have been demonstrated and are in use to varying degrees. The major categories are liquid crystal displays (LCD), plasma display panels (PDP), electroluminescent (EL), field emission displays (FED), vacuum fluorescent displays (VFD), digital micromirror devices (DMD), and light emitting diodes (LED). Table A-1 briefly describes each FPD technology. Although each technology has its own performance characteristics and is manufactured using different materials and processes, most are generally comprised of two glass plates surrounding a material that filters external light or emits its own light. These technologies use manufacturing techniques more similar to the production of semiconductor chips than televisions. Most FPDs control the color and brightness of each pixel (picture element) individually, rather than from one source, such as the electron gun in the CRT. The different types of electronic information display devices and how they are categorized are depicted in Fig. A-1.

1.1 Elimination of FPD Technologies from this Study

While there are several types of FPDs, two LCD technologies will be included in this LCA, based on their applicability to be used as substitutes in the computer display market. LCDs comprise approximately 87% of the FPD market (OTA, 1995). Currently, the largest market for FPDs is in notebook computers and CRTs monopolize the desktop computer market. However, FPDs are already moving into the desktop computer market. The LCD technology that best meets the purpose and needs of this study is the amorphous-silicon thin film-transitor (a:Si TFT) active matrix LCD (AMLCD). There are two variations of the a:Si TFT AMLCD that are expected to dominate the desktop monitor market for LCDs: the traditional twisted nematic (TN) mode and the in-plane switching (IPS) mode. Table A-1 describes these technologies. Various subtechnologies of LCDs are presented in Fig. A-2. The IPS mode is a non-nematic amorphous silicon AMLCD. Note that all the subtechnologies listed in Fig A-2 are not described here; the purpose is simply to show the complexity of different types of LCDs.

The PDP technology could be incorporated into the desktop computer market, especially if computers and televisions begin to merge. However, plasma technology is generally designed for large screens, and does not meet the specifications (e.g., diagonal size) of the functional unit defined for this project. Therefore, PDP technology will not be included in the scope of this project. FED and EL technologies are targeted toward military, medical, and high-end commercial products because they possess particular characteristics (such as size, durability, and high image quality) for those niche markets. Because these other FPD technologies are a small fraction of the market, not targeted toward the desktop computer market, and/or do not meet the

specifications of our functional unit, they are not included within the scope of this project. Table A-1 presents brief descriptions of various FPD technologies and whether or not they are included in this LCA.

Technology	Description	Applicability to Project
Liquid Crystal Displays (LCD)	A liquid crystal material, acting like a shutter, blocks, dims, or passes light unobstructed, depending on the magnitude of the electric field across the material (OTA, 1995). A backlight provides the light source.	Included in this study. Descriptions of the subtechnologies and whether or not they are included in the study are presented below.
(1) Passive matrix (PMLCD)	Liquid crystal (LC) material is sandwiched between two glass plates, which contain parallel sets of transparent electrical lines (electrodes) in a row and column configuration to form a matrix. Every intersection forms a pixel, and the voltage across the pixel causes the LC molecules to align and determines the shade of that pixel (OTA, 1995).	Traditionally for low-end applications (e.g., calculators, wrist watches). Higher end applications use a super-twisted nematic (STN) ¹ construction. The liquid crystal material is twisted between 180 and 270 degrees which improves the contrast between the "on" and "off" states, resulting in a clearer display than with the twisted nematic (twisted only 90 degrees) (OTA, 1995; MCC, 1997). However, cost and performance issues limit this technology from wide application in the desktop market and therefore, it will not be evaluated in this study.
(2) Active matrix (AMLCD)	Similar to the PMLCD except an electronic switch at every pixel provides faster switching and more shades. The addressing mechanism eliminates the viewing angle and brightness problems suffered by PMLCD. Requires more backlight than PMLCD due to the additional switching devices on the glass (at each pixel). Various switching types are listed below:	Provides vivid color graphics in portable computer and television screens (OTA, 1995). This technology meets the functional unit specifications in this study. Specific subcategories are described below.

Table A-1. Flat panel display technologies

¹ Traditional light modulating methods for LCD technologies include twisted nematic (TN), super-twisted nematic (STN), double STN, triple STN, and film-compensated STN (OTA, 1995). The STN is the current standard for high-end PMLCD applications.

Technology	Description	Applicability to Project
	AMLCD Switch Types:	
	(2a) Thin-film transistor (TFT): The transistor acts as a valve allowing current to flow to the pixel when a signal is applied. The transistors are made of various materials including: amorphous silicon (a:Si), polycrystalline silicon (p:Si), non-Si[CdSe] (Castellano, 1992). Two different TFT light modulating modes are twisted nematic (TN) and in-plane switching (IPS) (DisplaySearch 1998). In comparison to the TN mode, the IPS mode requires more backlight but fewer manufacturing steps.	The current standard AMLCD switching mechanism for computer displays is a:Si TFT. Polycrystalline Si is not suitable for larger than about 5" displays. Both the TN and IPS a:Si TFT AMLCD technologies are analyzed in this project.
	(2b) Diode matrix: The diode acts as a check valve. When closed, it allows current to flow to the pixel charging it. When opened, the pixel is disconnected and the charge is maintained until the next frame (Castellano, 1992).	The diodes are found to short easily and must be connected in series to achieve long life usability. The diode displays are also limited in size smaller than that of the functional unit.
	(2c) Metal-insulator metal (MIM): The MIM is a diode type switch using metal- insulated-metal fabrication techniques (OTA, 1995).	Temperature sensitive, which creates gray scale nonuniformities. They are also size limited like other diode type displays and therefore not included in this study.
(3) Active addressed LCD	Hybrid of passive and active matrix. The pixels are addressed using signals sent to the column and row as determined using an algorithm encoded into an integrated circuit (IC). The IC drives each row of pixels more or less continuously and drives multiple rows at one time (OTA, 1995)	Employed in notebook and desktop monitors >12.1". However, they need special drivers (OTA, 1995), have slow response times, and their contrast worsens as panel size increases (Young, 1998). Therefore, this technology does not meet the specifications of the functional unit and is excluded from evaluation in this study.
(4) Plasma-addressed liquid crystal (PALC)	The pixel is addressed using row electrodes, which send the signal, and column gas channels, which conduct a current when ionized (OTA, 1995).	PALC displays are in development to be used as large low cost displays. Production of the displays have not yet occurred and they are not included in this study.
(5) Ferroelectric LCDs (FLCD or FELCD)	The pixel is addressed using positive or negatives pulses to orient the crystals. The positive pulse allows light to pass (light state) and the negative pulse causes the blockage of light (dark state) (Castellano, 1992). A ferroelectric liquid crystal is bistable and holds it polarization when an electric field is applied and removed (Peddie, 1994). They are also called surface stabilized ferroelectric (SSF) LCD.	Has high resolution with very good brightness, but limited color palette (Peddie, 1994). Limited color palette does not meet color specification of functional unit.

Table A-1. Flat panel display technologies

Technology	Description	Applicability to Project
Plasma Display Panels (PDP)	An inert gas (e.g., He, Ne, Ar) trapped between the glass plates emits light when an electric current is passed through the matrix of lines on the glass. Glow discharge occurs when ionized gas undergoes recombination. Ionization of atoms occurs (electrons are removed), then electrons are recombined to release energy in the form of light. Full color plasma displays use phosphors that glow when illuminated by the gas (OTA, 1995).	Established technology. Good for large screens (e.g., wall-mounted televisions), but are heavier and require more power than LCDs (OTA, 1995). Designed for large screens and are larger displays than specified for desktop applications. Therefore, not included in this study.
Electroluminescent Displays (EL)	A phosphor film between glass plates emits light when an electric field is created across the film (OTA, 1995). EL uses a polycrystalline phosphor (similar to LED technology which is also an electroluminescent emitter, but uses a single crystal semi- conductor). ELs are doped (as a semiconductor) with specific impurities to provide energy states that lie slightly below those of mobile electrons and slightly above those of electrons bound to atoms. Impurity states are used to provide initial and final states in emitting transitions (Peddie, 1994). Also referred to as thin-film EL (AC-TFEL), active matrix EL (AMEL), DC EL, organic EL.	Lightweight and durable. Used in emergency rooms, on factory floors, and in commercial transportation vehicles (OTA, 1995). Problems found in the power consumption and controlling of gray levels. Targeted toward military, medical, and high-end commercial products, therefore not included in the scope of this project.
Field Emission Displays (FED)	Flat CRT with hundreds of cathodes (emitters) per pixel (form of cathodeluminescent display); eliminates single scanning electron beam of the CRT. Uses a flat cold (i.e., room temperature) cathode to emit electrons. Electrons are emitted from one side of the display and energize colored phosphors on the other side (OTA, 1995; Peddie, 1994).	Not commercially available, but anticipated to fill many display needs (OTA, 1995). Could potentially apply in all LCD and CRT applications. High image quality as with CRT, but less bulky and less power use than with CRT. A number of roadblocks to this technology taking over the AMLCD market include proven manufacturing processes (problems found in the reliability and reproducibility of the devices), efficient low-voltage phosphors, and high voltage drivers. The technology is targeted toward military, medical and high-end commercial products and not included in current study.
Vacuum Fluorescent Displays (VFD)	Form of cathodeluminescent display that employs a flat vacuum tube, a filament wire, a control grid structure, and a phosphor-coated anode. Can operate at low voltages since very thin layers of highly efficient phosphors are coated directly onto each transparent anode (Peddie, 1994).	VFDs offer high brightness, wide viewing angle, multi-color capability and mechanical reliability. Used in low information content applications (e.g., VCRs, microwaves, audio equipment, automobile instrument panels, etc.). No significant uses seen for computer displays (Peddie, 1994).

Table A-1.	Flat nanel	display	v technologies
	Fiat panti	uispia	icciniologics

Technology	Description	Applicability to Project
Digital Micromirror Devices (DMD)	Miniature array of tiny mirrors built on a semiconductor chip. The DMD is used in a projector that shines light on the mirror array. Depending on the position of a given mirror, that pixel in the display reflects light either onto a lens that projects it onto a screen (resulting in a light pixel) or away from the lens (resulting in a dark pixel) (OTA, 1995).	Just beginning to be used mainly as projection devices and has not been developed for use that would match the functional unit (OTA, 1995).
Light Emitting Diodes (LED)	The LED device is essentially a semiconductor diode, emitting light when a forward bias voltage is applied to a p-n junction. The light intensity is proportional to the bias current and the color dependent on the material used. The p-n junction is formed in a III-V group material, such as aluminum, gallium, indium, phosphorous, antimony, or arsenic.	For low information display applications, which makes it not capable of meeting the requirements of the functional unit. Color, power, and cost limitations prevent the emergence into the high information display market (Castellano, 1992).
Electrochromic display	Open-circuit memory using liquid electrolytes (Peddie, 1994, p. 214). Non-emitter (as LCDs), as opposed to emitters (e.g., EL, FED, PDP).	Outstanding contrast and normal and wide viewing angles; open-circuit memory. Complex and costly involving liquid electrolytes, poor resolution, poor cycle life, lack of multicolor capability, etc. Not suitable for computer displays in past; however, new technology may be promising (Peddie, 1994).
Light Emitting Polymers	Developing technology (Holton, 1997).	Developing technology.

Table A-1. Flat panel display techno	logies
--------------------------------------	--------



Figure A-1. Classification of Electronic Information Displays. *Source:* Adapted from Tannas 1985.



Figure A-2. LCD subtechnologies. Source: Catellano 1992.

this page intentionally left blank

APPENDIX B

DESCRIPTIONS OF LIQUID CRYSTAL DISPLAY (LCD) TECHNOLOGY AND AMORPHOUS SILICON THIN-FILM TRANSISTOR (a:Si TFT) TECHNOLOGY

this page intentionally left blank

PRINCIPLES OF LCD TECHNOLOGY



In this section, we will explain everything ranging from the properties of liquid crystal molecules to the basic principle of display technology by using TN type liquid crystals as an example.

When coming into contact with grooved surface in a fixed direction, liquid crystal molecules line up parallel along the

The parallel arrangement of liquid crystal molecules along grooves

OF LCD TECHNOLOGY

grooves.



AMORPHORU

PROJECTORS







grooves.

When liquid crystals are sandwiched between upper and lower plates, they line-up with grooves pointing in directions 'a' and 'b,' respectively



long axes parailei.

The molecules along the upper plate point in direction 'a' and those along the lower plate in direction 'b,' thus forcing the liquid crystals into a twisted structural arrangement. (figure shows a 90-degree twist) (TN type liquid crystal)

Light travels through the spacing of the molecular arrangement

The light also "twists" as it passes through the twisted liquid crystals

http://www.sharp-usa.com/products/pro/tech/principles.html

1/7/98



Light passes through liquid crystals, following the direction in which the molecules are arranged. When the molecule arrangement is twisted 90 degrees as shown in the tigure, the light also twists 90 degrees as it passes through the liquid crystals.

Molecules rearrange themselves when voltage is applied

When voltage is applied to the liquid crystal structure, the twisted light passes straight through.



The molecules in liquid crystals are easily rearranged by applying voltage or another external force. When voltage is applied, molecules rearrange themselves vertically (along with the electric field) and light passes straight through along the arrangement of molecules.

Blocking light with two polarizing filters

When voltage is applied to a combination of two polarizing filters and twisted liquid crystal, it becomes a LCD display.

http://www.sharp-usa.com/products/pro/tech/principles.html

1/7/98



Light passes when two polarizing filters are arranged with polarizing axes as shown above, left. Light is blocked when two polarizing filters are arranged with polarizing axes as shown above, right.

TN type LCDs

A combination of polarizing filters and twisted liquid crystal creates a liquid crystal display.



along perpendicular polarizing aces, light entering from above is re-directed 90 degrees along the helix arrangement of the liquid crystal molecules so that it passes through the lower filter.

This figure depicts the principle behind typical twisted nematic (TN) liquid crystal displays. In a TN type LCD, liquid crystals in which the molecules form a 90-degree twisted helix, are sandwiched between two polarizing filters. When no voltage is applied, light passes; when voltage is applied, light passes; when voltage is applied, light is blocked and the screen appears black. In other words, the voltage acts as a trigger causing the liquid crystals to function like the shutter of a camera.

9

FROM SHAPP MINDS

http://www.sharp-usa.com/products/pro/tech/principles.html



In 1988, Sharp developed a 14-inch TFT color TV, and with this development of a futuristic wall-mount TV, TFT LCDs created the foundation for manufacture and introduction of large-screen color displays.

Reinitzer discovered liquid crystals almost 100 years ago, and today, bolstered by customer needs and the new and special technologies and materials that a manufacturer can offer to meet those needs, liquid crystals have made huge strides.

At this opportunity, we would to make the whole world aware of the potential of LCDs, and as new manufacturers enter the market, become the trigger that raises this awareness to new levels.

In the evolution of LCD display manufacturing, the burden of undertaking aggressive development of application products has been considerable. In addition to notebook and sub-notebook PCs that have been the mainstream applications for LCDs in the past, there has been significant growth in areas which take advantage of the unique characteristics of LCD displays, such as compact size, thin profile, and low power consumption to create products which could not be produced using CRTs, such as LCD TVs, ViewCams, new portable information tools, etc. In addition, for large projection TVs, it has now become possible to develop products that are more compact and lighter in weight than conventional CRT-based models, and LCDs are rapidly becoming the mainstream display device in this field.

In this way, LCD displays have expanded into application areas that were once niches belonging solely to CRTs, and the development of numerous key technologies that have the potential to further expand their application product areas continues.

Thanks to the development of TFT LCD displays and the synergistic evolution (spiral evolution) with LCD application devices and equipment, such as PC notebooks and computer monitors, A/V equipment, car navigation systems, game devices, etc., we can anticipate the growth of new demand-generating products. LCDs have emerged as the likely winner among flat-panel displays for the new information-oriented society. As we approach the dawn of the multimedia era which will see the convergence of video, computers, and communications, a critical need is emerging for innovations in displays that link man and machine through our sense of sight.

The driving force behind LCD manufacturing are recently developed amorphous-Si TFT LCD technologies which represent breakthroughs in the areas of 1) higher aperture ratios, 2) wider viewing angles, and 3) EMI (electromagnetic interference) reduction, as well as low-temperature polycrystalline Si TFT LCD display technologies. Thanks to these

http://www.sharp-usa.com/products/pro/tech/tft.html

p,

breakthroughs, a new direction has emerged in 1996 which will make the best use of these key technologies in LCD applications. For example, higher aperture ratio technologies are being used in LCD displays intended for PC notebooks, wider viewing angle and lower EMI technologies are being used in LCDs destined for LCD monitors, and low-temperature polycrystalline Si TFT LCD technologies are being used as super-fine dot-pitch light valves for high-definition projection TV systems.

In the future, promising new technologies can be expected to spawn the next generation of new LCD application products based on high-performance LCD display systems ("systems-on-panel") that takes full advantage of integrated drive and control circuitry.

FROM SHARD MONDS

Ψ

http://www.sharp-usa.com/products/pro/tech/tft.html

1/7/98

References

Sharp. 1998. Information found on a Web page from Sharp USA. Web site available at: http://www.sharp-usa.com/products/pro/tech/.

APPENDIX C

CRITICAL REVIEW

this page intentionally left blank

APPENDIX C

CRITICAL REVIEW

Table C-1. Computer Display Project Core Group* and Technical Work Group Members

Contact	Contact Organization	
Salla Ahonen	Environmental Issues, Nokia Research Center, Nokia	Helsinki, Finland
Heather Bowman	Electronics Industry Alliance	Arlington, VA
Reggie Caudill	New Jersey Inst. of Tech. Research Center	Newark, NJ
Bob Donofrio	Display Device Consultants	Ann Arbor, MI
Holly Evans Co-Chair, Core Group	Electronics Industry Alliance	Arlington, VA
Bruce Gnade	DARPA	Arlington, VA
Tony Hainault	Minnesota Office of Environmental Assistance	St. Paul, MN
Kathy Hart* Co-Chair, Core Group	US EPA, Office of Prevention Pesticides and Toxic Substances	Washington, DC
Edwin Henderson*	US EPA, Office of Prevention Pesticides and Toxic Substances ^a	Washington, DC
David Isaacs* former Co-Chair, Core Group	Electronics Industry Alliance ^a	Arlington, VA
Mikko Jalas	Environmental Issues, Nokia Research Center, Nokia	Helsinki, Finland
Tim Jarvis	The SemiCycle Foundation	Austin, TX
Greg Keoleian	U. of Michigan, School of Natural Resources and the Env.	Ann Arbor, MI
Lori Kincaid*	Univ. of TN Center for Clean Products and Clean Technologies	Knoxville, TN
J. Ray Kirby	IBM	Research Triangle Park, NC
Jonathan Koch	GE Power Systems	Schenectady, NY
David Lear	Compaq Computer Corp.	Houston, TX
Clare Lindsay	U.S. EPA, Office of Solid Waste	Arlington, VA
John Lott	DuPont Electronic Materials	Research Triangle Park, NC
Jeff Lowry	Techneglas	Columbus, OH
Carole McCarthy	McCarthy Environmental Consulting	Duxbury, MA
Timothy Mann	IBM	Loganville, GA

APPENDIX C

Contact	Contact Organization	
Frank Marella Co-Chair, Technical Work Group	Sharp Electronics Corporation	Mahwah, NJ
John Mathews	Envirocycle	Hallstead, PA
Jay Mathewson	Eastman Kodak Co.	Rochester, NY
Colleen Mizuki*	Microelectronic & Computer Technology Corporation ^a	Austin, TX
Amanda Monchamp	Electronics Industry Alliance ^a	Arlington, VA
Rick Nolan	Motorola	Austin, TX
Bob Pinnel*	U.S. Display Consortium	San Jose, CA
Greg Pitts*	Microelectronic & Computer Technology Corporation ^a	Austin, TX
Gene Proch	Corning Asahi	
Gloria Schuldt	Microelectronic & Computer Technology Corporation ^a	Austin, TX
Eileen Sheehan	U.S. EPA, P2 Team, Region 9	San Francisco, CA
Dipti Singh* Co-chair, Technical Work Group	US EPA, Office of Prevention Pesticides and Toxic Substances	Washington, DC
Doug Smith	Sony Electronics Inc.	San Diego, CA
Ted Smith*	Silicon Valley Toxics Coalition	San Jose, CA
David Spengler*	Digital Equipment Corporation	Maynard, MA
Maria Socolof*	Univ. of Tennessee Center for Clean Products & Clean Technologies	Knoxville, TN
Dan Steele	Motorola MD FPD 10 ESIH and Chemical Operations Flat Panel Display Division	Tempe, AZ
Larry Stone	Compaq Computer Corp.	Houston, TX
Butch Teglas (Delmer F.)	Philips Consumer Electronics	Knoxville, TN
Valerie Thomas	Princeton Univ. Ctr. For Energy & Env. Studies	Princeton, NJ
David Thompson	Matsushita Electronic Corporation of America	Secaucus, NJ
Donna Timmons	Eastman Kodak Co.	Rochester, NY
Dani Tsuda*	Apple Computer Inc.	Cupertino, CA
Lucian Turk	Dell Computer	Austin, TX
Laura Turbini	Georgia Institute of Technology	Atlanta, GA
Victoria Wheeler	Eastman Kodak Co.	Rochester, NY

Table C-1. Computer Display Project Core Group* and Technical Work Group Members

APPENDIX C

Table C-1. Computer Display Project Core Group* and Technical Work Group Members			
Contact	Organization	Location	
Ross Young	Display Search	Austin, TX	
former Co-Chair,			
Technical Work Group			

* Core Group members (subset of Technical Work Group)

^a Affiliation at time of involvement in project.

Table C-2. U.S. EPA Design for the Environment Workgroup Members for the
Computer Display Project

Name	Division/Branch
Andrea Blaschka	Risk Assessment Division/Existing Chemicals Assessment Branch
Susan Dillman	National Program Chemicals Division/Technical Branch
Franklyn Hall	Economic, Exposure, and Technology Division/Chemical Engineering Branch
Kathy Hart	Economics, Exposure, and Technology Division/Design for the Environment
Karen Hogan	Risk Assessment Division/Science Support Branch
Susan Krueger	Economics, Exposure, and Technology Division/Economic and Policy Analysis Branch
Fred Metz	Economics, Exposure, and Technology Division/Industrial Chemistry Branch
Dipti Singh	Economics, Exposure, and Technology Division/Design for the Environment
Jerry Smrchek	Risk Assessment Division/Existing Chemicals Assessment Branch

this page intentionally left blank

APPENDIX D

TECHNICAL MEMORANDUM: Life-Cycle Inventory Approach for Materials Extraction and Materials Processing Life-Cycle Stages

this page intentionally left blank

APPENDIX D

TECHNICAL MEMORANDUM: Life-Cycle Inventory Approach for Materials Extraction and Materials Processing Life-Cycle Stages

1. INTRODUCTION

1.1 Background

The U.S. Environmental Protection Agency's Design for the Environment Program Computer Display Project (CDP) is conducting an environmental life-cycle assessment (LCA) that will evaluate the relative environmental impacts of cathode ray tubes (CRT) and liquid crystal display (LCD) computer monitors. The major life-cycle stages of a product system include materials extraction, materials processing, product manufacturing, product use, and final product disposition (end-of-life). An LCA evaluates the relative environmental impacts of a product system and is defined in greater detail in Chapter 1 of the main report. An LCA generally consists of four phases: goal definition and scoping, life-cycle inventory (LCI), lifecycle impact assessment (LCIA), and life-cycle improvement assessment.

The activity of quantifying the inputs (e.g., materials, utilities) and outputs (e.g., emissions, wastes) of a product system is the LCI phase of an LCA. A product system is made up of the multiple processes that help produce, use, or dispose of the product. Each process typically has an inventory that consists of inputs and outputs for each process. Therefore, an LCI of a product system consists of several inventories for processes throughout the life-cycle of the product. This technical memorandum (TM) addresses the LCIs related to two major life-cycle stages: materials extraction and materials processing, which together will be referred to as the life-cycle stages that are "upstream" of the product manufacturing stage. Ideally, transportation associated with those stages is also included. This TM will describe the approach to choosing the upstream data from secondary sources that will be included in the CDP analysis.

1.2 Purpose and Scope of this Technical Memorandum

The purpose of this TM is to present the approach for obtaining process-specific inventory data related to extraction and processing of the materials needed to produce a CRT and LCD computer monitor. Collecting these upstream inventory data can involve dozens of upstream processes because there are dozens of materials used to produce CRTs and LCDs. Therefore, decision rules are typically used to limit which materials to include in the scope of the LCA, and existing data from secondary sources are generally relied upon. For inventories related to materials extraction and materials processing, various databases with input and output LCI data exist for materials commonly used in industry. The existence of these inventories, and the limited resources available for collecting primary inventory data for the entire life cycle, result in the use of secondary data for product manufacturing and end-of-life processes. This TM identifies initial materials considered for inclusion in the upstream life-cycle stages. Actual material lists from the inventories collected from the primary data collection efforts were not available until

APPENDIX D

after data collection had to begin for the upstream processes. Therefore, initial materials were identified to help determine which secondary data to obtain. Once actual materials from the manufacturing stage inventory were identified, the selected secondary data source was checked for the appropriate data sets to be included in the study. This TM addresses the initial steps for choosing which upstream data source to use, by identifying and prioritizing several data sources. The remainder of this TM will present a brief summary of results, the methodology for selecting secondary upstream data for the CDP, detailed results in terms of preferred data sources, and the limitations to using the upstream data for the CDP.

2. **RESULTS SUMMARY**

Based on initial material lists and project decision rules, approximately 40 materials (including some material groups) were initially identified as materials for which upstream data inventories should be included in the CDP LCA. Nine data sources (i.e., studies and/or databases) were evaluated to determine which upstream data would be used for these and other materials that might be identified in the CDP. Two databases were disregarded because the data are not or will not be available to the public. The remaining seven were reviewed for their applicability to the CDP. Complete inventory data for all currently identified CDP materials were not available from any one of the databases/studies alone. Therefore, a hierarchy of preferred data has been chosen for upstream data from secondary sources. The most preferred data is that from the Environmental Information and Management Explorer (EIME) database developed by *Ecobilan* (Ecobalance), a company based in France.

EIME is an LCA software package that specializes in electronics and the electronics industry and currently includes 18, with forthcoming updates expected to bring it to 21 materials specific to the CDP. The database is immediately available, and although it is relatively expensive, it may be attainable at a negotiated price (Glazebrook 1999). The EIME data do not fulfill all the CDP's upstream data requirements and therefore, other databases will be needed. Twelve materials were not found in any of the databases and may require additional research from secondary or primary sources to complete the CDP product system inventories. It appears, however, that EIME, supplemented with Ecobalance's Database for Environmental Analysis and Management (DEAM) will cover most materials needed in the CDP.

3. METHODOLOGY

The method for determining the upstream data that will be used for the CDP depends on which materials need to be included in the upstream evaluation and what existing databases are currently available for those materials. This section consists of three subsections that present the following: (1) how the preliminary list of materials were identified; (2) which data sources were considered for use as CDP upstream inventory data; and (3) the selection criteria for choosing which upstream data to include in the CDP.

3.1 Materials Selection

The first step to selecting upstream data sources is to identify what materials are of interest to the project. Primary data collected from manufacturing facilities will provide a list of upstream materials to consider in the upstream stages. However, the materials inventory from

the CDP product manufacturing stage was not yet complete when upstream data collection needed to begin to meet project time and budget constraints. Therefore, a preliminary list of materials used to manufacture the monitors was identified by disassembling a CRT and LCD and by reviewing the literature on manufacturing processes. The list was then slightly reduced based on decision rules to limit the scope of the project. This preliminary list is then used to help choose preferred sources of upstream data for materials of interest in the CDP. The following subsections describe the bills of materials of the LCD and CRT, the decision rules applied to the bills of materials, and the list of selected materials for upstream data collection.

3.1.1 Bills of Materials

A 15" CRT and a 15" LCD desktop monitor were disassembled, to the extent they could be manually separated, into their component parts/materials and each of these parts was weighed using Mettler analytical balances. A 17" CRT (the CDP functional unit) was not available for disassembly and therefore it is assumed that the percent contribution of materials in the 15" CRT and the 17" CRT are equivalent, which is an adequate assumption for the purposes of identifying major product materials.

Primary (also referred to as "product") materials are defined as those that become part of the final assembled monitor. Bills of materials of the CRT and LCD monitors were compiled to quantify the mass contribution of each primary material and component in each monitor. Where individual materials could not be discerned, component parts consisting of multiple materials were identified and weighed. These bills of materials are presented in the CDP's Industry and Technology Profile Document (MCC 1998). The material makeup of some component parts [e.g., thin-film transistors (TFTs) on LCD glass substrate or phosphors on CRT glass substrate] were identified from published literature (i.e., secondary sources) (O'Mara 1993, DisplaySearch 1998, FCR 1996, MCC 1993, ECT 1980). Simultaneous and subsequent work on the CDP involved obtaining more details on the makeup of certain component parts from manufacturers (i.e., primary sources) through data collection questionnaires.

The next step was to identify common ancillary (also referred to as "process") materials used in product manufacturing, which were found from secondary sources (O'Mara 1993, DisplaySearch 1998, FCR 1996, MCC 1993, ECT 1980) and reviewed by industry experts. These ancillary materials were added to the primary bills of materials for consideration in the LCA (MCC 1998). Additional ancillary materials were identified from primary sources during concurrent manufacturing data collection activities.

3.1.2 Decision Rules

Due to the complexity of the CRT and LCD monitors, and for any LCA, the boundaries of the analysis must be clearly defined. Thus, the following decision rules for choosing the materials to be evaluated were developed and applied to the primary and ancillary bills of materials. Three major categories of decision criteria were used to select materials for detailed analysis in the LCA: (1) mass contribution; (2) potential environmental and/or energy significance; and (3) technological importance. A priority hierarchy was developed (Figure 1) using a combination of these criteria.



Figure 1. Decision rule hierarchy

The first criterion is applied by including materials that constitute greater than or equal to 1% of the monitor by mass. Materials constituting more than 5% will be given greater emphasis in the LCA. Mass is a simple measure by which to select important materials for consideration in the LCA because in many cases, the larger the material, the greater the impact. This is true for resource consumption impacts which are equivalent to the amount of material used. However, other impact categories may not be equivalent to the amount of material consumed, and simply eliminating materials based on mass alone may exclude important impacts from an environmental life-cycle perspective. Therefore, under the second criterion, materials were also included if they have a potential environmental/health impact (e.g., they may be toxic) or use large amounts of energy to produce. The environmental criterion decision rule refers to materials that may pose risks to the public, occupational workers, or the ecosystem from manufacturing, use, or disposal of the material. The primary and ancillary materials were reviewed by a team of experts at the University of Tennessee and were compared to regulatory lists and other sources (Klaassen et al. 1986, EPA 1998, ChemFinder 1998, SRC 1998) to identify materials with known or potential environmental concerns. When impacts are calculated in the LCIA, a more rigorous review of toxicity data and environmental parameters will be conducted to provide quantitative impact measures.

The third decision rule criterion applies to materials that are critical to the technology (e.g., LCD TFT materials or the CRT phosphors). This is intended to ensure that other materials of potential importance are not overlooked in the LCA. Furthermore, because the LCA will be comparative in nature, greater emphasis will be placed on materials that are physically unique to a display technology.

For the materials meeting the top tier of the decision rule hierarchy (Figure 1) in the CDP, attempts are made to obtain secondary data for those upstream material processes. Materials in the middle segment of the triangular hierarchy scheme are given lower priority, but included, if
available. Finally, the last segment of the triangle would contain materials excluded from the analysis.

3.1.3 Material Selection Results

The materials identified here are for selecting which materials require the collection of input and output inventory data from materials extraction, materials processing, and associated transportation, collectively referred to as the "upstream" life-cycle stages. The inventories from each of these life-cycle stages are then used to calculate impacts of the various impact categories considered in the analysis.

The total masses of the CRT and LCD that were disassembled were approximately 12.8 kg and 5.15 kg, respectively. The printed wiring boards (PWBs) and their components were excluded from these weights and from the following materials analysis because they are treated as complex display components not broken down by individual materials. The CRT consists of approximately 17 primary materials and the LCD is comprised of about 23 primary materials (MCC 1998). The major primary materials by weight ($\geq 1\%$) in the CRT and LCD are listed in Table 1 with their corresponding components. Figures 2 and 3 depict the percent contribution of each of those materials to the overall monitor. For the CRT, eight materials were greater than or equal to 1% and only three [glass, steel, and high impact polystyrene (HIPS)] were greater than 5% of the weight of the monitor. The LCD had seven materials greater than or equal to 1%, five of which were greater than 5% [steel, polycarbonate, acrylonitrile butadiene styrene (ABS), polyester, and glass]. The items in bold in Table 1 represent the materials that are >5% for both the CRT and LCD. Other primary materials to be included in the LCA, based on the environment and technology decision rules, are presented in Table 2. The primary materials that were excluded due to mass are presented in Table 3.

Ancillary materials, such as those required for photolithography, are used in greater quantities for LCDs than CRTs. Preliminary literature searches (O'Mara 1993, DisplaySearch 1998, FCR 1996, MCC 1993, ECT 1980) found four ancillary materials for CRTs and 12 for LCDs (MCC 1998). The latter portion of Table 2 presents the ancillary materials that are included for either technological or environmental importance. The mass criterion for ancillary materials will be identified through responses to data collection questionnaires distributed to manufacturers participating in the project. Table 3 shows the ancillary materials that were preliminarily excluded based on environmental and technical criteria because mass data for ancillary materials are not yet available.

Table 1. Primary materials comprising $\geq 1\%$ by mass of a CRT or LCD monitor and associated components ^a

Material	Associated component(s)			
	CRT	LCD		
ABS		Base/stand		
Aluminum (Al)	Aluminum shielding, power board heat sink, connectors	Power supply heat sink, TFT metal		
Copper (Cu)	Deflection yoke			
Ferrite-magnet	Deflection yoke			

Table 1. Primary materials comprising $\geq 1\%$ by mass of a CRT or LCD monitor and associated components ^a

Material	Associated of	component(s)	
	CRT	LCD	
Glass (e.g., borosilicate) ^b		LCP panel	
Glass (lead oxide)	Panel, funnel, neck, frit		
Lead (Pb) ^c	Funnel & neck glass, frit		
Plexiglas		Backlight clear protector	
Polycarbonate		Backlight light pipe	
Polyester		Power supply & rear cover insulators	
Polystyrene, high-impact (HIPS)	Casing		
Silicone	Potting material in flyback transformer		
Steel	Base, right, left & back shields; shadow mask	Base/stand weight & brackets, backlight plates, rear cover metal plate, power supply housing	

^a See Figures 2 and 3 for material percent contributions to total mass of monitor, excluding PWBs.

^b Includes materials that could not be easily separated from the glass (e.g., frit, phosphors, transistors) and subtracts the estimated lead content of the glass for the CRT.

^c The mass of lead was estimated from the total mass of the different glass components and approximate lead levels in the CRT glass components (MCC 1994). On average, approximately 10% of the total mass of CRT glass was assumed to be lead. NOTE: Materials in bold are >5% of the monitor by weight.





Table 2.	Primary and ancillary materials or components meeting technology (T),
	environment (Env), or energy (E) criteria

Materials Associated components or process			Decision	
			crit	eria
	CRT	LCD	CRT	LCD
Primary materials				
Aluminum oxide (Al_2O_3)	Electron gun wire heater		Env	
Aquadag	Faceplate black matrix		Т	
	coating			
Beryllium (Be)		Be-Cu metal clips		Env
Bismuth oxide	Shadow mask back		Env	
	coating			
Color filters (acryl epoxy resins)		Front panel glass color filters		Т
Divinylbenzene resin		Spacers in AMLCD cell		Env
Frit (lead solder glass)	Glass solder joints		Env, E	
Indium-tin oxide (ITO)		Electrode		Т
Liquid crystals (e.g., polycyclic aromatic halogenated		Light-modulating material		T, Env
hydrocarbons, cyanobiphenyl.				
phenylcyclohexane compounds)				
Mercury		Cold cathode fluorescent		Env
		tube in backlight		
Nickel	Electron gun cathodes		T, Env	
Phosphors (e.g., ZnS, Y_2O_2)	Illuminating material		T, Env	
Polyimide		AMLCD cell alignment		Т
		layer		
TFT metals (e.g., Al, Cr, Mo, W)		Transistor		T, Env
TFT silicon materials (e.g., SiO ₂ , SiNx, doped Si)		Transistor		Т
Tungsten (W)	Electron gun wire heater	Transistor	T, Env	T, Env
Ancillary materials				
Boron trichloride (BCl ₃)		Photolithographic etchant		Env
Carbon tetrafluoride (CF ₄)		Photolithographic etchant		Env
Carbon trifluoride (CHF ₃)		Photolithographic etchant		Env
Chloride (Cl ₂)		Photolithographic etchant		Env
Ferric chloride (FeCl ₃)	Photolithographic etchant (shadow mask)		Env	
Hydrochloric acid (HCl)		Photolithographic etchant		Env
Isopropyl alcohol (IPA)		Glass cleaner		Env
N-methyl pyrrolidone (NMP)		Photolithographic developer		Env
Polyvinyl alcohol	Photolithographic application of phosphors		Env	
Sulfur hexafluoride (SF ₆)		Photolithographic etchant		Env

Tetramethyl ammonium

hydroxide (TMAH)

environment (Env), or energy (E) criteria							
Materials	Associated components or process			Decision criteria			
	CRT	LCD	CRT	LCD			

Photolithographic

developer

Env

Table 2 Primary and ancillary materials or components meeting technology (T)

The materials identified for inclusion in the CDP (Tables 1 and 2) are then prioritized based on the decision rule hierarchy triangle. Those materials that are either: (1) > 5% by mass; or (2) of environmental/energy concern, fit into the top priority of the upstream data collection effort. Those materials that are either: (1) between 1-5% by mass; or (2) functionally important and/or physically unique, fit into a lower priority of upstream data collection, but are still included in the project. Those materials that are less than 1% by mass and do not meet the other criteria listed above are excluded from the analysis. Currently, the materials falling into each segment of the decision rule hierarchy are listed in Table 4.

Material	Associated component/application					
	CRT	LCD				
Primary materials						
Aluminized mylar		Corner tape on backlight assembly				
Brass	Brass ring on neck assembly	Brass threaded standoff in backlight assembly				
Foam rubber		Foam gasket in backlight assembly				
Nylon		Cable clamp, strain relief in backlight assembly, clamp in backlight, bushing in base/stand assembly				
Paper		Caution label on rear plate assembly				
Polysulphone	Insulating rings on neck assembly					
Silicone rubber		Gaskets in LCD panel assembly, shock cushion in light assembly, rubber feet in base/stand assembly				
Ancillary materials						
Nitrocellulose binder	For frit application					
Amyl acetate	For frit application					
0 ₂		Metal etchant				
N ₂		Metal etchant				
Iodine		Polarizer coating				

Table 3. Materials excluded from analysis

For the top priority materials, we obtained upstream inventory data from secondary sources where available. If no secondary sources were available, we attempted to collect primary data or conduct further research from the literature. For materials in the middle tier, we attempted to collect secondary data but gave less emphasis on including them if too many resources were required. The top tier consists of materials greater than 5% by mass and all the materials in Table 2. Each material in Table 2 was added to the list of materials for either environmental or technological reasons and all except tungsten (W) were unique to a technology. However, tungsten is also included in the top tier for potential environmental concern. As a result, all the materials in Table 2 are of potential environmental concern and/or are both functionally important and physically unique (see Figure 1).

Top tier	Middle tier	Lowest tier (excluded)
Steel, CRT glass, HIPS,	Ferrite-magnet, Silicone,	Aluminized mylar, Brass, Foam
Polycarbonate, ABS, Polyester,	Plexiglas, Al, Cu	rubber, Nylon, Paper,
LCD glass, lead, Al_2O_3 ,		Polysulphone, Silicone rubber,
Aquadag, Be, Bismuth oxide,		Nitrocellulose binder, Amyl
Acryl epoxy resins (color filters),		acetate, N_2 , O_2 , Iodine
Divinylbenzene resin, Frit, ITO,		
Liquid crystals, Hg, Ni,		
Phosphors, Polyimide, TFT		
metals, TFT silicon materials,		
W, BCl ₃ , CF ₄ , CHF ₃ , Cl ₂ , FeCl ₃ ,		
HCI, IPA, NMP, Polyvinyl		
alcohol, SF ₆ , TMAH		

 Table 4. Summary table of preliminary CDP materials in priority hierarchy

3.2 Data Sources Evaluated

In order to identify upstream inventory data to be used for the CDP, nine different data sources (databases or studies) were evaluated. The following nine were chosen based on UT's experience in LCA, which included a comprehensive review of LCA databases (Menke et al. 1996), and from the scoping process for this project:

• American Plastics Council (APC)

The APC is a major trade association for the U.S. plastics industry. APC is comprised of 24 of the leading plastics manufacturers in the United States with many members having a strong global market presence. APC's membership represents 80% of the U.S. resin production capacity (APC 1999). APC has collected LCI data that are expected to be released in 1999 for polyethylene (PE), polypropylene (PP), high impact polystyrene (HIPS), polyethylene terephthalate (PET), and polyvinyl chloride (PVC) resins and polyurethane precursors (Hentges 1999). Data are mostly vintage 1991 or 1993 and cover production in North America (Hentges 1999). Additional inventories from APC have not yet been identified, although they are presumed to exist.

APPENDIX D

•	Association of Plastics Manufacturors in Europa (ADME)
•	Association of Flashes Manufacturers in Europe (AFME)
	AF WE is an industry body that has published inventory data on otennis, polystytene (FS), DE DD DVC DET and nalymethanas (ADME 1000), as well as ADS, Disvision
	PE, PP, PVC, PET and polyineuranes (APME 1999), as well as ABS, Plexigias,
	polycarbonate, polyester, and polyimide (Karisson 1999).
•	Boustead
	Dr. Ian Boustead is a well known LCA practitioner who developed the Boustead model
	and database that allows users to produce LCIs of complete systems. Boustead's focus
	areas are aerosols, automotive products, beverage containers, building materials, and the
	plastics industry. The organization is based in the United Kingdom (Boustead 1999).
•	BUWAL
	BUWAL is the Swiss Agency for the Environment, Forests and Landscape (BUWAL
	1999). They have several published reports on LCA. BUWAL 250 is an English version
	of their LCI database of several common industrial materials.
•	Environmental Information and Management Explorer (EIME)
	This software design tool was developed by the <i>Ecobilan</i> group in conjunction with IBM,
	Alcatel, Legrand, Schneider, and Thompson. Ecobilan was founded in 1990 and has
	offices in Europe and in the United States (<i>Ecobilan</i> 1999). Version 1.4 of EIME has
	been released and the embedded database contains 170 modules on the most commonly
	used materials and subcomponents of the electronic and electric industry (EIME 1999).
•	Industrial DEsign MATerials (IDEMAT)
	Dr. J.A.M. Remmerswaal and J. Rombouts of the Delft University of Technology's
	Section for Environmental Product Development produced this software with a database
	of LCI data for various industrial materials. There is a student version that was released
	in 1995 that is available to the public at no cost, but the availability and cost of the
	complete version is yet undetermined by UT. This evaluation focuses on the student
	version that was available to UT.
•	New Jersey Institute of Technology (NJIT) Report
	NJIT's "Lifecycle Assessment of Television CRTs," (Caudill 1998) report is not a
	database of upstream inventory data per se, however, it is a preliminary LCA that
	includes LCI data for a CRT and therefore it was considered for use in the CDP as an
	upstream data source.
•	Personal Computer (PC) Ecolabel Report
	This study was developed by Atlantic Consulting and IPU (Institute for Product
	Development of the Technical University of Denmark) for the Ecolabel Unit of the
	European Commission (AC and IPU 1998). The purpose of the report was to study
	personal computers so that an ecolabel could possibly be established. Similar to the NJIT
	report, this is an LCA with inventory data applicable to the CDP, but it is not a traditional
	database of upstream inventory data. Analysis of the inventory in the PC Ecolabel Report
	for the purposes of the CDP was based on Version 1.11 of the report downloaded from
	their website in January of 1998.
•	United States Automotive Materials Partnership (USAMP)
	Formed in June 1993, this partnership set out to conduct vehicle-oriented research and
	development in materials and materials processing to improve the competitiveness of the
	Lice is the discharge the second by the seco

development in materials and materials processing to improve the competitiveness of the U.S. auto industry (USAMP 1999). The USAMP is conducting joint research to further the development of lightweight materials for improved automotive fuel economy. The major technology groups being studied are polymer composites, light metals (including

aluminum, magnesium, etc.), engineered plastics, cast iron, steel and ceramics (USAMP 1999). The aluminum, plastics, steel and automotive industries are participating in a collaborative LCI project to produce a quantitative database of information regarding all the resources used to make, operate and dispose of a generic 3200-pound vehicle (USAMP 1999).

3.3 Selection Criteria

For choosing which upstream data to use for the CDP, the following 11 criteria were considered. Descriptions of each criterion and what is preferred for that category are presented below:

1. **Geographic boundaries** – Describes whether the data are representative of Europe and/or the United States. In general, U.S. data are preferred for this project, assuming most of the materials for the monitors are extracted and processed in the U.S. However, because some of the CDP manufacturing is in Asia, materials may originate from non-U.S. locations/countries.

2. Origin of data – Describes whether or not the data originate from primary or secondary sources. Primary sources that are clearly identified are preferred.

3. Currency of data – This refers to the dates that represent the actual inventory data. More recent data are preferred. If the date of the inventory data is not known, the date the database was released is considered.

4. Public availability – Data are categorized as either public or private. Publicly available data can be considered for the CDP.

5. When available – This describes whether or not the data are currently available and if not, when they are expected to be available. Immediately available means the data are available from the appropriate company or individual; however, more time may be required for UT to obtain the data. Also provided under this criterion will be whether UT currently has some of the data applicable to the CDP. Immediately available is preferred and further consideration is given to data that UT already has in house.

6. **Cost** – Due to limited resources, cost is an important factor for determining which upstream data should be obtained for the CDP. However, if possible, negotiations for or donations of data can be pursued as this is a collaborative project with industry and other stakeholders. Least costly data are preferred.

7. Upstream life-cycle stages – Which upstream life-cycle stages are included from each data source are identified, if possible. In some cases, the databases or reports address more than only upstream stages and other stages included will also be noted under this criterion. In the results of this analysis, we will present the names of the life-cycle stages as they are labeled in each respective data source. However, such labels may not be consistent with the specific

terminology used in this report. For use in the CDP, we prefer data sources that include materials extraction, materials processing, and associated transportation.

8. Aggregation of data – This describes whether or not the data from the various life-cycle stages are aggregated into one set of inventory numbers or how the data are aggregated. With less aggregation, the CDP will better be able to predict impacts particular to a specific life-cycle stage. Therefore, less aggregation is preferred. For some of the reports considered in this analysis, processes may also be aggregated for an entire product or component and therefore it is difficult to separate out the inventory for one particular material. The advantage of material-specific LCI databases is that the data are not aggregated into a larger component or product.

9. Input/output categories – This lists which categories of inputs and outputs are included in the database or report (e.g., non-renewable resources, fuel and energy inputs, water use, air emissions, water effluents, solid/hazardous wastes). Ideally, the input and output categories would match those defined for the CDP that will be used to calculate the impacts. The LCIA TM (Socolof 1999) describes the impact categories and how the inventory data will be used to calculate impacts. Also of interest is whether the outputs within each category are chemical specific. The more speciated the chemicals, the more desirable the data. In some cases, chemical groups or categories of chemicals are provided. The CDP LCIA methodology requires chemical-speciated data to calculate most impacts.

10. Data quality indicators – If the data source provides an indication of its data quality, this will help determine the data quality of the CDP. In several cases, we were not able to discern whether there were data quality indicators for a particular data set. If the data source provides some indication of data quality, this can then be incorporated into the CDP's data quality indicators for the upstream data.

11. CDP materials included – These are the materials that have been identified in Set. 3.1.3, which constitute the initial list of materials of interest in the CDP. They were cross-referenced with each data source under consideration. Preferred data sources are those with the greatest number of materials of interest to the CDP.

Each database or report was reviewed based on available information, and in some cases, limited information was available. This exercise was not intended to be a comprehensive review of each database, because we were not able to purchase each source. It was intended to be a cursory review of available data sources to assist in the decision of which inventory data to obtain and include in the CDP for the upstream life-cycle stages. When we could not obtain a database, our review was based on information available on company websites, other available literature on the database, personal contacts with company representatives, or third parties who have had experience with using a particular database.

Based on all this information, preferred data sources were identified. All factors were considered, including expected data quality and cost. The first most important criterion was whether the data source included many of the materials of interest to the CDP. Although additional materials may be identified during the concurrent CDP data collection efforts, we expect that the majority of materials of interest have already been identified.

4. **RESULTS**

Using the CDP decision rules, we initially identified approximately 40 materials (including some material groups) for which upstream data inventories should be included in the CDP LCA (Sect. 3.1.3). Nine data sources were evaluated to determine which upstream data would be used for these and other materials that might be identified. Tables 5 and 6 present a comparison of the data sources evaluated. Table 5 lists the first ten criteria presented in Sect. 3.3, which are related to the type of data provided, availability of the data, and cost. Table 6 cross-references the materials of interest in the CDP to the materials found in the various data sources (the 11th criterion in Sect. 3.3). Together this information was used to identify which data are preferred for use in the CDP as upstream inventory data. Brief discussions of each data source and a final conclusion are presented below.

Referring to Table 5, APC data are not yet available and it is uncertain if they will be available as scheduled, as they were expected to be released in previous years but were not. Therefore, APC is not considered further in this analysis. USAMP inventory data, which were intended only for participating organizations is not a publicly available data set. Therefore, USAMP as a source of upstream data for the CDP is also excluded from further analysis. The remaining seven data sources are evaluated by analyzing Tables 5 and 6.

The eleven criteria described above (Sect. 3.3) can be condensed into three major areas:

- Cost;
- Data quality; and
- Applicability to CDP.

Each source will be described in terms of these criteria, without giving a particular weight to any one over another. Note that the "data quality" criterion is a combination of the origin of the data, the currency of the data, the upstream life-cycle stages included, data quality indicators, and to some extent, the geographic boundaries of the data (see Table 5). The "applicability to the CDP" criterion depends on which upstream life-cycle stages are included, how the data are aggregated, what input and output categories are included (including whether or not the output data are speciated), whether data quality indicators are provided, and which materials of interest to the CDP are included.

	APC	APME	Boustead	BUWAL 250	EIME	IDEMAT	NJIT LCA	PC Ecolabel	UPAMP
Geographic boundaries	U.S.	Europe	Europe	Europe	U.S. & Europe	Netherlands & Europe	U.S.	Europe	U.S.
Origin of data	Unknown	Primary	Primary	Secondary	Majority is primary, some secondary	Unknown	Secondary	Secondary	Unknown
Currency of data	1990s	1990s (varies per material)	Unknown	1996	1990s (varies)	Second Student version, released in 1995	1970s - 1990s	Not completely determined, but most appear to be 1990s	1990s
Public availability	Public	Public	Public	Public	Public	Public (student version); unknown for complete version	Public	Public	Public
When available	Expected to be released in 1999	Immediate (UT has 2 applicable materials)	Immediate	Immediate (UT has 6 applicable materials)	Immediate	Immediate (UT has 9 applicable materials; unknown if complete version can be obtained)	Interim report immediately available (UT has copy)	Version 1.11 immediately available (UT has copy)	Not available to public
Cost	No cost	No cost	~\$10,000	~\$250	~\$7,500; > \$5,700 for universities (Negotiable)	No cost, unknown for complete version	No cost	Version 1.11 no cost; ~\$75 for final report	Not available to public
Upstream life-cycle stages	Unknown	Raw material extraction, material processing, transport	Process operations (including fuel production) and transport operations	Pre- combustion, combustion + processes, transports	Extraction, processing, transportation	Production, which includes transportation when noted, and not clear if extraction included	Material extraction and material synthesis	Material production, manufacturing, transport, use, EOL	Unknown

 Table 5. Selected criteria of upstream data sources

	APC	APME	Boustead	BUWAL 250	EIME	IDEMAT	NJIT LCA	PC Ecolabel	UPAMP
Aggregation of data	Unknown	Some data presented as process- specific e.g., fuel production, transport, process)	Often classified into several processes: fuel production, fuel use, transport operations, process operations	Aggregated as "LCI" or "energy consumption," latter subclassified (e.g., final energy source, energy supply, final process energy, transport)	Aggregated over all upstream life- cycle stages for each module (material) into impact categories, system administration can access LCI data separately	Each material aggregated for all life-cycle stages for the following categories: processes, thermal energy, electrical energy, and transports	Process specific; sometimes a few subprocesses are aggregated	Aggregated by major computer components (e.g., monitor) for each life-cycle stage, not process or material specific	Unknown
Input/ Output categories	Unknown	Energy, primary fuels, and raw material inputs; air, water, and solid waste emissions; outputs mostly unspecified	Gross energy, primary fuels & feedstocks: raw materials; water use; air, water solid waste emissions; outputs provided as chemical categories and several speciated abamicals	Commercial fuels resources, feedstock resources, materials used in final stage, main product, co-products, usable wastes, waste treatment; outputs provided as chemical categories and as some speciated abamiagh	Natural resources, energy, water inputs; air, water, hazardous waste outputs; outputs relatively well speciated	Material inputs (including water), energy inputs; air, water, and solid outputs; mostly unspeciated outputs	Raw material and energy inputs; solid, air, and waterborne waste outputs; outputs as chemical categories and some speciation	Resource consumption (raw materials), air emissions, water emissions, and waste; includes chemical categories, but also very well speciated	Unknown

Table 5. Selected criteria of upstream data sources

Table 5.	Selected	criteria o	f upstream	data sources
----------	----------	------------	------------	--------------

	APC	APME	Boustead	BUWAL 250	EIME	IDEMAT	NJIT LCA	PC Ecolabel	UPAMP
Data	Unknown	All	Not	Unknown	Provides high,	Unknown	Data were gathered	Unknown	Unknown
quality		calculations	provided, but		medium, and		on each material,		
indicators		were	data quality		low measures		carefully citing		
		referred	believed to		of reliability of		notes and references		
		back to	be		the data		which document the		
		participatin	moderately				original sources		
		g companies	good (above						
		before	average as						
		being used	compared to						
			other						
			available						
			sources)						

Key:

APC = American Plastics Council

APME = Association of Plastics Manufacturers in Europe

EIME = Environmental Information and Management Explorer

IDEMAT = Industrial DEsign MATerials

NJIT = New Jersey Institute of Technology

USAMP = U.S. Automotive Materials Partnership

4.1 Data Source Reviews

APME data are European-based, of moderate quality, and available for free. However, the data are mostly limited to materials of interest to the plastics industry and therefore only apply to 7 materials of interest to the CDP. The materials covered by APME constitute 17% of the product weight of the CRT described in Figure 2 and 49% of weight of the LCD in Figure 3.

The Boustead data are very expensive, of moderate quality, and include several materials of interest for the CDP, including most of the major product materials by weight of the CRTs and LCDs (55% and 85%, respectively, see Table 6). The significant material missing for the CRT is the leaded glass, which is approximately 39% of the mass of the monitor. The Boustead data include two of the ancillary materials that have been identified for the CDP, but do not include several of the other materials identified for potential environmental concern. The Boustead data are moderately equipped with speciated chemical data as required for the CDP. Data are aggregated for all upstream stages, but are also available as some individual upstream inventories (e.g., transport operations, process operations). Although data quality indicators are not provided by Boustead for the data, the Center for Clean Products and Clean Technologies assesses it as above average based on comparisons with other databases reviewed.

		APME	Boustead	BUWAL 250	EIME ⁴	IDEMAT ⁸	NJIT	PC Ecolabel
Prir	nary Materials							
1	ABS	Y	Y		Y (APME)	Y	Y	Y
2	Aluminum		Y	Y	Y	Y	Y	Y
3	Aluminum oxide		Y	Y	Y (BUWAL)			
4	Aquadag							
5	Beryllium					Y		
6	Bismuth oxide					Y (C) ⁹		
7	Color filters (acryl epoxy resins)				Y ⁵			
8	Copper		Y		Y	Y	Y	Y
9	Chromium (TFT metal)		mining chromite ore		(Y)	Y		
10	Divinylbenzene resin							
11	Ferrite-magnet		Y ²		Y ⁶			Y
12	Frit							Y
13	Glass, borosilicate (LCD)	Y	Y	Y	(Y)	Y		
14	Glass, lead oxide (CRT)						Y	Y
15	Indium-tin oxide - ITO							
16	Lead		Y		Y	Y	Y	Y
17	Liquid crystals				Y			
18	Mercury					Y (C)		
19	Molybdenum (TFT metal)				(Y)	Y		
20	Nickel				Y	Y		Y

 Table 6. Cross-reference of preliminary CDP materials and potential upstream data sources

		APME	Boustead	BUWAL 250	EIME ⁴	IDEMAT ⁸	NJIT	PC Ecolabel
21	Phosphors: e.g., ZnS, Y ₂ O ₂							
22	Plexiglas [polymerization of methyl ester (methyl methacrylate)]	Y			Y (APME)			
23	Polyimide	Y			Y (APME)			
24	Polycarbonate	Y	Y		Y (APME)	Y	Y	Y
25	Polyester	Y			Y (APME)	Y (C)		
26	Polystyrene-HIPS	Y	Y	Y	Y (APME)	Y	Y	Y 11
27	Silicon TFT materials: e.g., SiNx, SiO2				Y	Y (C)		
28	Silicone				Y			
29	Steel		Y	Y	Y(BUWAL)	Y (C)	Y	Y
30	Tungsten (TFT metal)					Y		Y
Anc	illary Materials							
31	Boron trichloride							
32	Carbon tetrafluoride							
33	Chlorine		Y	Y				
34	Ferric chloride							
35	Hydrochloric acid		Y		Y			
36	Isopropyl alcohol							
37	N-methyl pyrrolidone							
38	Polyvinyl alcohol							
39	Surfur hexafluoride							
40	Tetramethyl ammonium hydroxide							
Tota	als	7	12 ³	6	18 (21) 7	12 (17) 10	8	12
% co mate	ontribution of CRT primary erials $\geq 1\%$ by mass ¹	17	55	46	56	25 (52) 10	91	94
% co mate	ontribution of LCD primary priats $\geq =$ by mass ¹	49	85	57	88 (97) ⁷	38 (96) ¹⁰	76	76

Table 6. Cross-reference of preliminary CDP materials and
potential upstream data sources

1. The percent mass contribution of the primary materials were summed to identify the total percent of the monitor by mass that is covered by each data source.

2. Inventory data are available for iron, which is assumed to represent ferrite-magnet.

3. The tally of chemicals for Boustead excludes mining chromite ore for chromium.

4. Cells with a Y and the name of a data source [e.g., "Y (APME)"] indicate the data source that EIME obtained that particular inventory from, if that source is included elsewhere in this table. "(Y)" represents materials that are expected in EIME's forthcoming update.

5. "Epoxy resins" assumed to be for color filters (acryl epoxy resins).

6. The EIME database does not have ferrite-magnet listed, but it does have ferrites MnZn as a mateiral inventory.

7. The first value represents the current EIME dataset and the value in parenthesis indicates materials expected in the forthcoming update.

8. The student version of IDEMAT was the source investigated. "C" (for "complete") indicates cases where IDEMAT has the inventory on a given material only in the complete version.

9. Bismuth

10. The first value is for the student version and the value in parthenthesis is for the complete version.

11. The study only listed polystyrene (PS) and did not indicate if it was high-impact polystyrene (HIPS).

BUWAL 250 is a relatively inexpensive database that is also European data and believed to be from secondary sources. Therefore, the data quality is marginal and its applicability to the CDP is also relatively low as it only appears to cover the least number of materials of interest. Some chemical speciation is provided as output data and the inventories are aggregated over the upstream processes.

EIME appears to be the best candidate for the purposes of the CDP as it is targeted specifically for the electronics industry and includes many of the materials of interest to the CDP (Table 6). The current version includes 18 materials and, with forthcoming updates to the database, there are expected to be 21 materials covered (Karlsson 1999), representing 56% of the materials by mass of the CRT and 97% of the LCD. The low CRT percent is again due to the lack of leaded glass, which is 39% of the CRT. These data also include important materials not included in the weight criterion such as liquid crystals, lead, silicon materials, and color filters. Each of these, with the exception of lead are not found in any other data sources reviewed. The EIME inventory output data appear to be relatively well speciated, but the inventory data in general cannot be separated into each upstream life-cycle stage. The data quality is adequate as some data are from the U.S. and from primary sources. The data specific to the electronics components are from the five industrial partners and their suppliers, while some of the other common industrial materials were obtained from other LCI databases (EIME 1999). The cost is relatively high; however it covers the cost of the entire life-cycle software tool. Discussions with *Ecobilan* representatives have revealed their willingness to negotiate for the use of some material inventory data, provided we supply our results to them in a desirable format. Alternatively, we would be required to purchase the entire software package to obtain the desired inventory data.

The student version of IDEMAT is another free set of data that is European-based. The true quality of the data is not yet well determined by UT. Several (12) CDP materials are included in the student version, and it is believed that 17 would be covered with the complete version. IDEMAT has a few materials, metals in particular, that are not found in any of the other data sources reviewed.

The NJIT LCA report provides only eight material inventories relevant to the CDP; however, because it is an LCA of a television with a CRT, it includes leaded glass, which is not commonly found in existing databases. Some of the inventory data, which are U.S.-based, are from relatively old secondary sources (*circa* 1970). Furthermore, not all outputs are quantified. The report includes two upstream life-cycle stages: materials extraction and "materials synthesis" (referred to as "materials processing" in the CDP). Transportation within these upstream stages is not included. Data are easily identified per material, and some chemical speciation is included. This report does not provide sufficient amount of upstream data to be used exclusively, but given that it is available at no cost, it may supplement missing data (e.g., leaded glass).

The PC Ecolabel LCA is a report that includes 12 of the materials of interest in the CDP in its study and has very well speciated output data. UT has obtained a copy of Version 1.11 at no charge. It is based on European data and of undetermined quality. This was a study that was intended to present results of the life-cycle impacts of a PC and is not intended to be a database of material inventories. UT chose to evaluate this as a potential source for upstream data because of the relevant subject matter of the LCA. And although it covers several materials of interest, the inventory data cannot be separated into individual material inventories. Data are presented for different life-cycle stages, but not provided on a material basis. Therefore, this report could be helpful for checking our final results of the LCA, but not for providing upstream inventories of specific materials.

4.2 Conclusion

To identify the priorities for using upstream data, we would prefer to use as much data from one source as possible to help ensure consistency and thus improve the data quality of the results of the CDP. We have selected to target EIME as the primary source of upstream data. However, what is seen from the information in Table 6 is that no one data source will completely encompass the product materials of the CDP. Twelve of the 40 materials were not covered by any data source. EIME includes the greatest number, yet even including the expected updates to EIME, ten primary materials and nine ancillary materials are not covered by EIME. Four of the ten primary materials and eight of the nine ancillary materials were not covered by any other data sources. Six materials that are not in EIME are believed to be in other data sources. These include the following: beryllium, bismuth oxide, frit, leaded glass, mercury, and tungsten. Beryllium and tungsten are in the student version of IDEMAT and bismuth and mercury are in the complete version of IDEMAT. The NJIT and PC Ecolabel studies include leaded glass and the PC Ecolabel study also includes frit. The NJIT report, however, does not quantify the outputs from leaded glass production. Furthermore, because the PC Ecolabel study's inventory is aggregated over the whole monitor, individual inventory data cannot be produced for the specific materials. Therefore, IDEMAT data and the quantified inputs from the NJIT report may help supplement the EIME data.

Subsequent work for the CDP revealed that Ecobalance also had DEAM data available that supplemented the EIME data, both of which were listed as upstream data sources for this project. Procuring EIME requires negotiations with *Ecobilan* for a reduced price. This will begin subsequent to the final approval of this TM by EPA and the CDP Core and Technical Work Groups. In the event we cannot procure the EIME data for a reduced price, we will try to rely on the no cost options of APME, IDEMAT and NJIT data. Together, these three data sources cover 17 materials with the IDEMAT student version or 20 with the complete version. Relying on several material inventories from each of the three sources will reduce consistency in our upstream data and thus reduce the data quality in the CDP. For materials not included in the data sources obtained, UT will attempt to find the data from primary or secondary sources.

5. LIMITATIONS AND UNCERTAINTIES

Information on the different databases were from personal communications, websites, and in some cases, review of selected inventories obtained for a database. Copies of the NJIT and Ecolabel reports were available for review. This evaluation of upstream sources was not intended to be a comprehensive assessment of each data source, but instead a cursory review to evaluate which data to pursue. Therefore, there remain uncertainties to the information presented in this TM, but it is believed that adequate information was available to make recommendations in this report.

Using secondary data will also have an effect on the limitations of the CDP results. Using secondary data that are not tailored to the specific goals and boundaries of a project limits the quality of the data. However, due to the large data collection efforts in the LCA, priorities are given to collecting data. Thus, secondary sources have been chosen for upstream inventories and primary sources will be approached for monitor and component manufacturing data, as well as some end-of-life processes. Once the upstream data are incorporated into the CDP, limitations in those databases will be transferred to limitations in the CDP. Furthermore, the use of more than one database (e.g., EIME supplemented by NJIT and IDEMAT) will reduce consistency in our upstream data and thus somewhat reduce the data quality in the CDP. However, it should be noted that the upstream data are only one portion of the overall inventory of the product systems being evaluated.

ACRONYMS/ABBREVIATIONS

ABS	=	Acrylonitrile butadiene styrene
APC	=	American Plastics Council
APME	=	Association of Plastics Manufacturers in Europe
BUWAL	=	The Swiss Agency for the Environment, Forests and Landscape
CDP	=	Computer Display Project
CRT	=	Cathode ray tube
DEAM	=	Database for environmental Analysis and Management
EIME	=	Environmental Information and Management Explorer
IDEMAT	=	Industrial DEsign MATerials
IPA	=	Isopropyl alcohol
IPU	=	Institute for Product Development of the Technical University of Denmark
HIPS	=	High impact polystyrene
LCA	=	Life-cycle assessment
LCD	=	Liquid crystal display
LCI	=	Life-cycle inventory
LCIA	=	Life-cycle impact assessment
LDPE	=	Low density polyethylene
NJIT	=	New Jersey Institute of Technology
NMP	=	N-methyl pyrrolidone
PC	=	Personal computer
PS	=	Polystyrene
PE	=	Polyethylene
PET	=	Polyethylene terephthalate
PP	=	Polypropylene
PVC	=	Polyvinyl chloride
PWB	=	Printed wiring board
TFT	=	Thin-film transistor
ТМ	=	Technical memorandum
TMAH	=	Tetramethyl ammonium hydroxide
USAMP	=	United States Automotive Materials Partnership

REFERENCES

- APC (American Plastics Council). 1999. Web site available at http://www.ameriplas.org/apcorg/about apc/about apc.html.
- APME (Association of Plastic Manufacturers in Europe). 1999. Web site entitled, "Life Cycle Assessment (LCA). A Guide to Approaches, Experiences and Information Sources," under heading, "Sort by Organisation Name," available at http://tiger.eea.eu.int/projects/EnvMaST/lca/kap451.htm.
- AC (Atlantic Consulting) and IPU (Institute for Product Development of the Technical University of Denmark). 1998. *LCA Study of the Product Group Personal Computers in the EU Ecolabel Scheme*, Version 1.11. January.
- Boustead. 1999. Web site entitled, "Life Cycle Assessment (LCA). A Guide to Approaches, Experiences and Information Sources," under heading, "Sort by Organisation Name," available at http://tiger.eea.eu.int/projects/EnvMaST/lca/kap451.htm.
- BUWAL. 1999. Web site entitled, "Life Cycle Assessment (LCA). A Guide to Approaches, Experiences and Information Sources," under heading, "Sort by Organisation Name," available at http://tiger.eea.eu.int/projects/EnvMaST/lca/kap451.htm.
- Caudill, R. J. 1998. Interim Report from the Multi-lifecycle Engineering and Manufacturing Program: Lifecycle Assessment of Television CRTs. Multi-lifecycle Engineering Research Center, New Jersey Institute of Technology. Newark, NJ. May 20.
- ChemFinder. 1998. CS ChemFinder web site available at chemfinder/camsoft.com, downloaded December, 1998.
- DisplaySearch. 1998. *DisplaySearch FPD Equipment and Materials Analysis and Forecast*. Austin, TX. June.
- *Ecobilan.* 1999. Web site entitled, "Life Cycle Assessment (LCA). A Guide to Approaches, Experiences and Information Sources," under heading, "Sort by Organisation Name," available at http://tiger.eea.eu.int/projects/EnvMaST/lca/kap451.htm.
- ETC. 1980. Encyclopedia of Chemical Technology, 3rd Edition, vol. 11, 1980, p. 847.
- EIME. 1999. Web site available at http://www.ecobalance.com/software/eime/eime ovr.htm.
- EPA (Environmental Protection Agency). 1998. Web site available at http://www.epa.gov, downloaded December.
- FCR (Fuji Chimera Research). 1996. The Future of Liquid Crystal and Related Display Materials, p. 12, translated by InterLingua Linguistic Services, Inc. Redondo Beach, CA 1997.

- Glazebrook, B. 1999. Personal communication with B. Glazebrook, *Ecobilan* (Ecobalance), and M. L. Socolof, University of Tennessee. May.
- IDEMAT. 1995. Delft University of Technology, Industrial DEsign MATerials software.
- Hentges, S. 1999. Personal communication with S. Hentges, American Plastics Council, and L. Lasher, University of Tennessee. February 1.
- Karlsson, M. 1999. Personal communication with M. Karlsson, IBM Engineering Center for Environmentally Conscious Products, and M. L. Socolof, University of Tennessee. March 1 and July 8.
- Klaassen, C. D., M. O. Amdur, and J. Doull. 1986. *Casarett and Doull's Toxicology The Basic Science of Poisons, 3rd Edition*. Macmillan Publishing Co., New York, NY.
- Menke, D., G. A. Davis, and B. W. Vignon 1996. *Evaluation of Life-Cycle Assessment Tools*. University of Tennessee, Knoxville, TN. August 30.
- MCC (Microelectronics and Computer Technology Corporation). 1993. Environmental Consciousness: A Strategic Competitiveness Issue for the Electronics and Computer Industry. Austin, TX.
- MCC. 1994. Electronics Industry Environmental Roadmap. Austin, TX
- MCC. 1998. Computer Display Industry and Technology Profile. EPA 744-R-98-005.
- O'Mara, W. C. 1993. "Liquid Crystal Flat Panel Displays." *Manufacturing Science and Technology*. New York, NY.
- Socolof, M. L. 1999. Draft Technical Memorandum Life-Cycle Impact Assessment Methodology for the Computer Display Project. University of Tennessee. March 25.
- SRC (Syracuse Research Corporation). 1998. Web site available at esc.syrres.com, downloaded December.
- USAMP (United States Automotive Materials Partnership). 1999. Web site available at http://www.uscar.org/index.htm.

APPENDIX E

TECHNICAL MEMORANDUM: Electrical Energy Grid Life-Cycle Inventories for the CDP

this page intentionally left blank

APPENDIX E

TECHNICAL MEMORANDUM: Electrical Energy Grid Life-Cycle Inventories for the CDP

1. INTRODUCTION

1.1 Background

The U.S. Environmental Protection Agency's (EPA) Design for the Environment (DfE) Program Computer Display Project (CDP) is conducting a combined Cleaner Technologies Substitutes Assessment (CTSA) and life-cycle assessment (LCA) to evaluate the relative environmental impacts, cost, and performance of cathode ray tube (CRT) and active matrix liquid crystal display (AMLCD) desktop computer monitors. Initially the project is conducting an LCA to determine the relative potential environmental impacts of the monitors, including impacts from materials extraction, materials processing, manufacturing (monitor and components), use, and end-of-life disposition. Transportation information is also included within and between each stage.

LCA has four major components:

- 1. goal definition and scoping,
- 2. life-cycle inventory (LCI),
- 3. life-cycle impact assessment (LCIA), and
- 4. improvement assessment.

The LCI is a collection of inputs (materials, energy, and other resources) and outputs (products, wastes and emissions) for processes throughout the product's life cycle. The LCIA characterizes the potential relative impacts of these inputs and outputs per functional unit, where the functional unit is defined in this project as one monitor over its lifetime. Improvement assessment is the component where those who are in the position to make changes in either the design or manufacture of the product review the results and decide on ways to implement environmental improvements.

The focus of this technical memorandum (TM) is on the LCI component of the LCA. Established LCI methodology accounts for electricity requirements throughout the life cycle of a product system. Wherever electricity is used in a process in the product system, the LCI typically includes the inputs and outputs from the generation of that electricity. This TM presents the input and output inventory data that are used to calculate the impacts from electricity generation for the CDP. The inventory data were developed by the University of Tennessee Center for Clean Products and Clean Technologies (CCPCT) from existing data for the various fuels (e.g., coal, petroleum) used by electric utilities around the United States. Two inventories illustrate the amount of materials consumed (inputs) and pollutants released (outputs) to generate one kilowatthour (kWh) of electricity, based on the average U.S. and the average Japanese electrical grids. The data are presented in units of grams of material input or emission output per kWh (g/kWh) in almost all cases, excluding radioactive emissions which are presented in Becquerels per kWh [Bq/kWh, where a Becquerel is the Système Internationale (SI) unit for radioactivity]. All of the inputs and outputs in the electric grid inventories have been multiplied by the electricity consumption rate for each process in the product system where electricity is used. These input and output data are not used with LCI data collected from secondary sources that already include inputs and outputs from electricity generation. The two electric grid inventory data sets presented in this TM are for the average U.S. electrical grid and the average Japanese electric grid.

1.2 Boundaries of the Analysis

In addition to the geographic boundary (i.e., U.S. and Japanese data only), the boundaries of the data presented in this TM are defined by the generation categories included in the analysis (e.g., coal, nuclear) and the life-cycle stages evaluated for the various generation categories (e.g., extraction, generation). A U.S. electric grid inventory was developed first and the Japanese grid inventory is simply a modification of the U.S. inventory, which applies the average Japanese electric grid to the input and output data found for the United States. This will not account for any effects of different control technologies used in Japan compared to the United States. The majority of this TM focuses on collecting U.S. data, and mentions when the Japanese grid is applied.

Table 1 presents the electricity generation categories, associated fuels types, percent breakdown for U.S. electricity generation in 1997 and whether or not that category is included in this inventory. (While 1998 is the target year for the overall CDP, this TM was targeted for 1997 as almost no data could be obtained that was relative to 1998.) Coal is the dominant generation category and fuel type in the U.S., accounting for over 55 percent of 1997 electricity production. Non-renewable fuels (coal, gas, petroleum, and uranium) plus water (for hydroelectric plants) together provide greater than 99 percent of U.S. electricity. Table 1 also lists the average Japanese breakdown of fuels.

Generation Category	Fuel	U.S. Generation % Breakdown	Japanese % Breakdown	Included in Electric Grid Analysis?
Coal	Coal	57.23%	18%	Yes
Gas	Gas	9.07%	20%	Yes
Petroleum	Petroleum	2.53%	21%	Yes
Nuclear	Uranium	20.14%	31%	Yes
Hydro	Water	10.79%	9%	No
Renewables	Wind, biomass, heat from sun, heat from earth	0.24%	1%	No
Total		100%	100%	

Table 1. Fuel Types Used to Generate Electricity in U.S. and Japan in 1997

Sources: EIA 1999a, EIA 1997.

Individual input and output inventories were developed for each of the generation categories listed in Table 1, except hydro and renewables. Hydroelectric facilities, which constitute nearly 11% of the U.S. electric grid, were excluded due to the scarcity of data on hydroelectric inputs and outputs. Known impacts of hydroelectric facilities primarily relate to

reservoir formation, which includes habitat destruction and its concurrent effects on biodiversity, and the generation of the greenhouse gases methane and carbon dioxide from the flooding of wetlands during and after reservoir formation. Habitat destruction is not included as an impact category *per se* within the CDP LCIA methodology (see Chapter 3 of main report), but global warming is included. Furthermore, EPA has concluded there is currently no adequate basis for estimating the emissions and sinks from the flooding of wetlands (EPA 1998a).

Renewables were excluded because they accounted for only a small fraction (0.24%) of total U.S. electricity production in 1997. In addition, little or no data exist on material inputs and pollutant outputs for renewable electricity generation processes.

The following life-cycle stages are associated with electricity generation:

- Extraction of ores and fluids from the earth and any necessary transportation to the initial processing point (Materials Extraction);
- Initial and secondary processing of those ores/fluids into usable fuels, and the associated transportation between processing points and to the generating stations (Materials Processing);
- Combustion or use of the fuel and the onsite control of pollutants, and transmission and distribution of the generated electricity to the points of use (Manufacturing);

NOTE: This description excludes the process flows of renewable energy sources.

Little or no data were available for the extraction and initial and secondary processing of ores and fluids into fuels, except for some data for coal extraction and processing. Therefore, most of the electrical grid input and output data presented in this TM only include the manufacturing life-cycle stage (generation and transmission and distribution of electricity) as shown in Table 2.

Life-Cycle Substages	Coal	Gas	Petroleum	Uranium
Extraction	Р	N	N	N
Transport to Initial Processing	N	N	N	Ν
Initial Processing	Р	Ν	N	Ν
Transport to secondary processing	NA	NA	NA	Ν
Secondary processing	NA	NA	NA	Ν
Transport to generating stations	Ν	Ν	N	Ν
Electricity generation	Ι	Ι	Ι	Ι
Transmission and Distribution	Ι	Ι	Ι	Ι

 Table 2. Fuel Life-Cycle Substages Captured in Electric Grid LCI Data

I = Included in inventory; P = Partially included in inventory; N = Not included in inventory; NA = Not applicable.

It should also be noted that this electric grid inventory relates only to *utility*-based generation. Electric power generation in the Unites States can be broken down into two main categories: utility and nonutility. In simplified terms, most electricity generating entities that are not classified as utilities fall in the nonutility category, and include many cogeneration facilities and small and independent power producers. About 65% of nonutility production is attributed to

the manufacturing sector (EIA 1998a), and nonutility production as a whole constitutes just over 10% of the total power produced by the electric power industry. The inputs and outputs of nonutility electricity production are excluded here due to a lack of detailed data on the associated emissions and wastes.

1.3 Organization of the TM

The remainder of this document provides supporting information on the development of the electric grid data. It is organized into six sections: results summary, methodology, fuel-specific results, data sources and quality, limitations and conclusions. Supporting tables are presented as necessary in the Appendices.

2. **RESULTS SUMMARY**

Table 3a presents the primary and ancillary materials consumed and products and environmental burdens produced during the generation of 1 kilowatthour (kWh) in the United States, based on the national 1997 generation percent breakdown (see Table 1). Similarly, Table 3b presents the inventory for the Japanese grid. All inputs and outputs listed in Tables 3a and b are presented on a per kWh basis.

Section 3 of this TM discusses the methods used to calculate the nationwide U.S. electric grid inventory data. Spreadsheets were used to organize and manipulate all of the inventory data, and those spreadsheets are shown in Attachments A through E and contain supporting data source, assumptions, and limitations information.

Material	Quantity	Units	Input/output	Input/output type	Disposition
PRIMARY INPUTS					
Coal, avg. (in ground)	2.83E+02	G	Input	Primary material	
Natural gas	2.20E+01	G	Input	Primary material	
Petroleum (in ground)	5.99E+00	G	Input	Primary material	
Uranium, yellowcake	7.64E-03	G	Input	Primary material	
ANCILLARY INPUTS					
Lime	1.67E+00	G	Input	Ancillary material	
Limestone	3.79E+00	G	Input	Ancillary material	
Water	1.79E+03	G	Input	Water	
PRODUCT					
Electricity	1.00E+00	KWH	Output	Energy	
AIR EMISSIONS					
1,1,1-Trichloroethane	3.02E-06	G	Output	Airborne	air
1,2-Dichloroethane	5.65E-06	G	Output	Airborne	air
2,3,7,8-TCDD	2.02E-12	G	Output	Airborne	air
2,3,7,8-TCDF	7.20E-12	G	Output	Airborne	air
2,4-Dinitrotoluene	3.96E-08	G	Output	Airborne	air
2-Chloroacetophenone	9.89E-07	G	Output	Airborne	air
2-Methylnaphthalene	4.20E-09	G	Output	Airborne	air
5-Methyl chrysene	3.11E-09	G	Output	Airborne	air
Acenaphthene	8.94E-08	G	Output	Airborne	air
Acenaphthylene	3.55E-08	G	Output	Airborne	air
Acetaldehyde	8.05E-05	G	Output	Airborne	air
Acetophenone	2.12E-06	G	Output	Airborne	air
Acrolein	4.10E-05	G	Output	Airborne	air
Anthracene	3.07E-08	G	Output	Airborne	air
Antimony	6.87E-06	G	Output	Airborne	air
Arsenic	5.91E-05	G	Output	Airborne	air
Barium	3.24E-06	G	Output	Airborne	air
Benzene	1.84E-04	G	Output	Airborne	air
Benzo[a]anthracene	1.46E-08	G	Output	Airborne	air
Benzo[a]pyrene	5.37E-09	G	Output	Airborne	air
Benzo[b,j,k]fluoranthene	1.68E-08	G	Output	Airborne	air
Benzo[g,h,i]perylene	5.68E-09	G	Output	Airborne	air
Benzyl chloride	9.89E-05	G	Output	Airborne	air
Beryllium	3.01E-06	G	Output	Airborne	air
Biphenyl	2.40E-07	G	Output	Airborne	air
Bromoform	5.51E-06	G	Output	Airborne	air
Bromomethane	2.26E-05	G	Output	Airborne	air
Cadmium	7.59E-06	G	Output	Airborne	air

Table 3a. U.S. Electricity Generation Inventory [inputs and outputs per kWh (3.6 MJ)]

Material	Quantity	Units	Input/output	Input/output type	Disposition
Carbon dioxide	7.00E+02	G	Output	Airborne	air
Carbon disulfide	1.84E-05	G	Output	Airborne	air
Carbon monoxide	1.27E-01	G	Output	Airborne	air
Chloride ions	2.86E-04	G	Output	Airborne	air
Chlorobenzene	3.11E-06	G	Output	Airborne	air
Chloroform	8.33E-06	G	Output	Airborne	air
Chromium (III)	3.83E-05	G	Output	Airborne	air
Chromium (VI)	1.14E-05	G	Output	Airborne	air
Chrysene	1.61E-08	G	Output	Airborne	air
Cobalt	1.91E-05	G	Output	Airborne	air
Copper	1.57E-06	G	Output	Airborne	air
Cumene	7.49E-07	G	Output	Airborne	air
Cyanide (-1)	3.53E-04	G	Output	Airborne	air
Di(2-ethylhexyl)phthalate	1.03E-05	G	Output	Airborne	air
Dibenzo[a,h]anthracene	1.38E-09	G	Output	Airborne	air
Dichloromethane	4.10E-05	G	Output	Airborne	air
Dimethyl sulfate	6.78E-06	G	Output	Airborne	air
Dioxins, remaining unspeciated	9.21E-11	G	Output	Airborne	air
Ethyl Chloride	5.93E-06	G	Output	Airborne	air
Ethylbenzene	1.33E-05	G	Output	Airborne	air
Ethylene dibromide	1.70E-07	G	Output	Airborne	air
Fluoranthene	1.06E-07	G	Output	Airborne	air
Fluorene	1.32E-07	G	Output	Airborne	air
Fluoride	3.08E-05	G	Output	Airborne	air
Formaldehyde	1.33E-04	G	Output	Airborne	air
Furans, remaining unspeciated	1.47E-10	G	Output	Airborne	air
Hexane	9.46E-06	G	Output	Airborne	air
Hydrochloric acid	1.70E-01	G	Output	Airborne	air
Hydrofluoric acid	2.12E-02	G	Output	Airborne	air
Indeno(1,2,3-cd)pyrene	1.04E-08	G	Output	Airborne	air
Isophorone	8.19E-05	G	Output	Airborne	air
Lead	2.01E-05	G	Output	Airborne	air
Magnesium	1.55E-03	G	Output	Airborne	air
Manganese	7.18E-05	G	Output	Airborne	air
Mercury	1.18E-05	G	Output	Airborne	air
Methane	1.02E+00	G	Output	Airborne	air
Methyl chloride	7.49E-05	G	Output	Airborne	air
Methyl ethyl ketone	5.51E-05	G	Output	Airborne	air
Methyl hydrazine	2.40E-05	G	Output	Airborne	air
Methyl methacrylate	2.83E-06	G	Output	Airborne	air

Table 3a. U.S. Electricity Generation Inventory [inputs and outputs per kWh (3.6 MJ)]

Material	Quantity	Units	Input/output	Input/output type	Disposition
Methyl tert-butyl ether	4.94E-06	G	Output	Airborne	air
Molybdenum	9.20E-07	G	Output	Airborne	air
Naphthalene	2.88E-06	G	Output	Airborne	air
Nickel	1.08E-04	G	Output	Airborne	air
Nitrogen oxides	1.85E+00	G	Output	Airborne	air
Nitrous oxide	5.35E-03	G	Output	Airborne	air
o-xylene	8.99E-08	G	Output	Airborne	air
PM-10	9.10E-02	G	Output	Airborne	air
Phenanthrene	3.95E-07	G	Output	Airborne	air
Phenol	2.26E-06	G	Output	Airborne	air
Phosphorus (yellow or white)	7.80E-06	G	Output	Airborne	air
Propionaldehyde	5.37E-05	G	Output	Airborne	air
Pyrene	5.25E-08	G	Output	Airborne	air
Selenium	1.84E-04	G	Output	Airborne	air
Styrene	3.53E-06	G	Output	Airborne	air
Sulfur dioxide	3.93E+00	G	Output	Airborne	air
TOCs, remaining unspeciated	9.07E-03	G	Output	Airborne	air
Tetrachloroethylene	6.07E-06	G	Output	Airborne	air
Toluene	4.00E-05	G	Output	Airborne	air
Vanadium	2.77E-05	G	Output	Airborne	air
Vinyl acetate	1.07E-06	G	Output	Airborne	air
Xylene (mixed isomers)	5.23E-06	G	Output	Airborne	air
Zinc (elemental)	2.40E-05	G	Output	Airborne	air
WATER RELEASES					
Sulfate ion (-4)	1.08E-01	G	Output	Waterborne	surface water
Suspended solids	2.80E-03	G	Output	Waterborne	surface water

Table 3a.	U.S. Electricity	Generation	Inventory [inputs and	outputs p	er kWh (3.6 MJ)]
	•/		•/ L					

Material	Quantity	Units	Input/output	Input/output type	Disposition
WASTE	•	•			•
Coal waste	8.01E+01	G	Output	Solid waste	landfill
Dust/sludge	3.10E+01	G	Output	Solid waste	landfill
Fly/bottom ash	2.00E+01	G	Output	Solid waste	landfill
Low-level radioactive waste	2.77E-03	G	Output	Radioactive waste	landfill
Uranium, depleted	8.30E-04	G	Output	Radioactive waste	landfill
RADIOACTIVE AIR EMISSION	S	-	-		
Argon-41 (isotope)	2.51E+01	Bq	Output	Radioactivity	air
Bromine-89 (isotope)	2.91E-06	Bq	Output	Radioactivity	air
Bromine-90 (isotope)	1.18E-06	Bq	Output	Radioactivity	air
Cesium-134 (isotope)	7.99E-05	Bq	Output	Radioactivity	air
Cesium-137 (isotope)	6.02E-04	Bq	Output	Radioactivity	air
Chromium-51 (isotope)	1.58E-03	Bq	Output	Radioactivity	air
Cobalt-57 (isotope)	4.24E-06	Bq	Output	Radioactivity	air
Cobalt-58 (isotope)	5.41E+00	Bq	Output	Radioactivity	air
Cobalt-60 (isotope)	4.07E-04	Bq	Output	Radioactivity	air
Iodine-131 (isotope)	1.90E-03	Bq	Output	Radioactivity	air
Iodine-132 (isotope)	3.86E-04	Bq	Output	Radioactivity	air
Iodine-133 (isotope)	1.76E+00	Bq	Output	Radioactivity	air
Iodine-134 (isotope)	2.00E-03	Bq	Output	Radioactivity	air
Iodine-135 (isotope)	1.01E-04	Bq	Output	Radioactivity	air
Krypton-85 (isotope)	4.17E+01	Bq	Output	Radioactivity	air
Krypton-85M (isotope)	2.02E+00	Bq	Output	Radioactivity	air
Krypton-87 (isotope)	7.52E-01	Bq	Output	Radioactivity	air
Krypton-88 (isotope)	3.53E+00	Bq	Output	Radioactivity	air
Manganese-54 (isotope)	2.24E-05	Bq	Output	Radioactivity	air
Niobium-95 (isotope)	8.89E-07	Bq	Output	Radioactivity	air
Rubidium-88 (isotope)	8.26E-03	Bq	Output	Radioactivity	air
Silver-110M (isotope)	2.65E-08	Bq	Output	Radioactivity	air
Technetium-99M (isotope)	1.19E-07	Bq	Output	Radioactivity	air
Tritium-3 (isotope)	5.90E+01	Bq	Output	Radioactivity	air
Xenon-131M (isotope)	3.40E+00	Bq	Output	Radioactivity	air
Xenon-133 (isotope)	4.91E+02	Bq	Output	Radioactivity	air
Xenon-133M (isotope)	3.26E+01	Bq	Output	Radioactivity	air
Xenon-135 (isotope)	1.85E+01	Bq	Output	Radioactivity	air
Xenon-135M (isotope)	3.54E-01	Bq	Output	Radioactivity	air
Xenon-138 (isotope)	1.17E+00	Bq	Output	Radioactivity	air

 Table 3a. U.S. Electricity Generation Inventory [inputs and outputs per kWh (3.6 MJ)]

Material	Quantity	Units	Input/output	Input/output type	Disposition
Zirconium-95 (isotope)	2.30E-06	Ba	Output	Radioactivity	air
RADIOACTIVE WATER RELE	ASES	24	ourput		un
Antimony-124 (isotope)	1.24E-02	Bq	Output	Radioactivity	surface water
Antimony-125 (isotope)	4.95E-02	Bq	Output	Radioactivity	surface water
Barium-140 (isotope)	9.21E-04	Ba	Output	Radioactivity	surface water
Cesium-134 (isotope)	3.32E-02	Bq	Output	Radioactivity	surface water
Cesium-136 (isotope)	3.84E-14	Bq	Output	Radioactivity	surface water
Cesium-137 (isotope)	4.99E-02	Bq	Output	Radioactivity	surface water
Chromium-51 (isotope)	5.98E-02	Bq	Output	Radioactivity	surface water
Cobalt-57 (isotope)	1.45E-03	Bq	Output	Radioactivity	surface water
Cobalt-58 (isotope)	5.90E-01	Bq	Output	Radioactivity	surface water
Cobalt-80 (isotope)	1.55E-01	Bq	Output	Radioactivity	surface water
Iodine-131 (isotope)	2.76E-02	Bq	Output	Radioactivity	surface water
Iodine-132 (isotope)	1.05E-02	Bq	Output	Radioactivity	surface water
Iodine-133 (isotope)	1.18E-02	Bq	Output	Radioactivity	surface water
Iodine-135 (isotope)	8.49E-03	Bq	Output	Radioactivity	surface water
Iron-55 (isotope)	1.41E-01	Bq	Output	Radioactivity	surface water
Iron-59 (isotope)	7.24E-03	Bq	Output	Radioactivity	surface water
Krypton-85M (isotope)	3.73E-02	Bq	Output	Radioactivity	surface water
Lanthanum-140 (isotope)	9.86E-04	Bq	Output	Radioactivity	surface water
Manganese-54 (isotope)	3.94E-02	Bq	Output	Radioactivity	surface water
Molybdenum-99 (isotope)	7.44E+04	Bq	Output	Radioactivity	surface water
Niobium-95 (isotope)	1.02E-02	Bq	Output	Radioactivity	surface water
Ruthenium-103 (isotope)	1.24E-03	Bq	Output	Radioactivity	surface water
Silver-110M (isotope)	1.45E-02	Bq	Output	Radioactivity	surface water
Sodium-24 (isotope)	2.21E-03	Bq	Output	Radioactivity	surface water
Strontium-89 (isotope)	2.39E-03	Bq	Output	Radioactivity	surface water
Strontium-90 (isotope)	5.61E-04	Bq	Output	Radioactivity	surface water
Strontium-95 (isotope)	6.18E-03	Bq	Output	Radioactivity	surface water
Sulfur-136 (isotope)	1.33E-03	Bq	Output	Radioactivity	surface water
Technetium-99M (isotope)	8.66E-04	Bq	Output	Radioactivity	surface water
Tin-113 (isotope)	1.37E-03	Bq	Output	Radioactivity	surface water
Tritium-3 (isotope)	4.41E+02	Bq	Output	Radioactivity	surface water
Xenon-131M (isotope)	4.54E-01	Bq	Output	Radioactivity	surface water
Xenon-133 (isotope)	6.97E+01	Bq	Output	Radioactivity	surface water
Xenon-133M (isotope)	5.71E-01	Bq	Output	Radioactivity	surface water
Xenon-135 (isotope)	5.20E-01	Bq	Output	Radioactivity	surface water

Table 3a. U.S. Electricity Generation Inventory [inputs and outputs per kWh (3.6 MJ)]

Table 3a. C.S. Electricity Generation Inventory [inputs and outputs per kwin (3.6 (13)]								
Material	Quantity	Units	Input/output	Input/output type	Disposition			
Zinc-85 (isotope)	6.65E-04	Bq	Output	Radioactivity	surface water			

Table 3a. U.S. Electricity Generation Inventory [inputs and outputs per kWh (3.6 MJ)]

Table 3b. Japanese Electricity Generation Inventory [inputs and outputs per kWh (3.6 MJ)]

Material	Quantity	Units	Input/output	Input/output type	Disposition			
PRIMARY INPUTS								
Coal, avg. (in ground)	8.88E+01	G	Input	Primary material				
Natural gas	4.85E+01	G	Input	Primary material				
Petroleum (in ground)	5.00E+01	G	Input	Primary material				
Uranium, yellowcake	1.18E-02	G	Input	Primary material				
ANCILLARY INPUTS								
Lime	5.24E-01	G	Input	Ancillary material				
Limestone	1.19E+00	G	Input	Ancillary material				
Water	1.72E+03	G	Input	Water				
PRODUCT		-						
Electricity	1.00E+00	KWH	Output	Energy				
AIR EMISSIONS		-						
1,1,1-Trichloroethane	2.51E-06	G	Output	Airborne	air			
1,2-Dichloroethane	1.78E-06	G	Output	Airborne	air			
2,3,7,8-TCDD	6.53E-13	G	Output	Airborne	air			
2,3,7,8-TCDF	2.27E-12	G	Output	Airborne	air			
2,4-Dinitrotoluene	1.24E-08	G	Output	Airborne	air			
2-Chloroacetophenone	3.11E-07	G	Output	Airborne	air			
2-Methylnaphthalene	9.24E-09	G	Output	Airborne	air			
5-Methyl chrysene	9.77E-10	G	Output	Airborne	air			
Acenaphthene	1.68E-07	G	Output	Airborne	air			
Acenaphthylene	1.28E-08	G	Output	Airborne	air			
Acetaldehyde	2.53E-05	G	Output	Airborne	air			
Acetophenone	6.66E-07	G	Output	Airborne	air			
Acrolein	1.29E-05	G	Output	Airborne	air			
Anthracene	1.77E-08	G	Output	Airborne	air			
Antimony	3.69E-05	G	Output	Airborne	air			
Arsenic	2.72E-05	G	Output	Airborne	air			
Barium	2.01E-05	G	Output	Airborne	air			
Benzene	5.92E-05	G	Output	Airborne	air			
Benzo[a]anthracene	3.11E-08	G	Output	Airborne	air			
Benzo[a]pyrene	1.69E-09	G	Output	Airborne	air			
Benzo[b,j,k]fluoranthene	1.51E-08	G	Output	Airborne	air			
Benzo[g,h,i]perylene	1.67E-08	G	Output	Airborne	air			
Benzyl chloride	3.11E-05	G	Output	Airborne	air			
Beryllium	1.26E-06	G	Output	Airborne	air			

Material	Quantity	Units	Input/output	Input/output type	Disposition
Biphenyl	7.55E-08	G	Output	Airborne	air
Bromoform	1.73E-06	G	Output	Airborne	air
Bromomethane	7.11E-06	G	Output	Airborne	air
Cadmium	5.48E-06	G	Output	Airborne	air
Carbon dioxide	5.98E+02	G	Output	Airborne	air
Carbon disulfide	5.78E-06	G	Output	Airborne	air
Carbon monoxide	1.09E-01	G	Output	Airborne	air
Chloride ions	2.39E-03	G	Output	Airborne	air
Chlorobenzene	9.77E-07	G	Output	Airborne	air
Chloroform	2.62E-06	G	Output	Airborne	air
Chromium (III)	2.15E-05	G	Output	Airborne	air
Chromium (VI)	5.22E-06	G	Output	Airborne	air
Chrysene	2.08E-08	G	Output	Airborne	air
Cobalt	4.60E-05	G	Output	Airborne	air
Copper	1.24E-05	G	Output	Airborne	air
Cumene hydroperoxide	2.35E-07	G	Output	Airborne	air
Cyanide (-1)	1.11E-04	G	Output	Airborne	air
Di(2-ethylhexyl)phthalate	3.24E-06	G	Output	Airborne	air
Dibenzo[a,h]anthracene	1.15E-08	G	Output	Airborne	air
Dichloromethane	1.29E-05	G	Output	Airborne	air
Dimethyl sulfate	2.13E-06	G	Output	Airborne	air
Dioxins, remaining unspeciated	2.90E-11	G	Output	Airborne	air
Ethyl Chloride	1.87E-06	G	Output	Airborne	air
Ethylbenzene	4.61E-06	G	Output	Airborne	air
Ethylene dibromide	5.33E-08	G	Output	Airborne	air
Fluoranthene	6.79E-08	G	Output	Airborne	air
Fluorene	7.11E-08	G	Output	Airborne	air
Fluorides (F-)	2.57E-04	G	Output	Airborne	air
Formaldehyde	3.97E-04	G	Output	Airborne	air
Furans, remaining unspeciated	4.62E-11	G	Output	Airborne	air
Hexane	2.98E-06	G	Output	Airborne	air
Hydrochloric acid	5.33E-02	G	Output	Airborne	air
Hydrofluoric acid	6.66E-03	G	Output	Airborne	air
Indeno(1,2,3-cd)pyrene	1.74E-08	G	Output	Airborne	air
Isophorone	2.58E-05	G	Output	Airborne	air
Lead (Pb, ore)	1.71E-05	G	Output	Airborne	air
Magnesium	4.89E-04	G	Output	Airborne	air
Manganese (Mn, ore)	4.24E-05	G	Output	Airborne	air
Mercury	4.59E-06	G	Output	Airborne	air

Table 3b. Japanese Electricity Generation Inventory [inputs and outputs per kWh (3.6 MJ)]

Material	Quantity	Units	Input/output	Input/output type	Disposition		
Methane	3.16E-03	G	Output	Airborne	air		
Methyl chloride	2.35E-05	G	Output	Airborne	air		
Methyl ethyl ketone	1.73E-05	G	Output	Airborne	air		
Methyl hydrazine	7.55E-06	G	Output	Airborne	air		
Methyl methacrylate	8.88E-07	G	Output	Airborne	air		
Methyl tert-butyl ether	1.55E-06	G	Output	Airborne	air		
Molybdenum	6.01E-06	G	Output	Airborne	air		
Naphthalene	8.60E-06	G	Output	Airborne	air		
Nickel	5.72E-04	G	Output	Airborne	air		
Nitrogen oxides	1.58E+00	G	Output	Airborne	air		
Nitrous oxide	4.34E-03	G	Output	Airborne	air		
o-xylene	7.50E-07	G	Output	Airborne	air		
Phenanthrene	2.02E-07	G	Output	Airborne	air		
Phenol	7.11E-07	G	Output	Airborne	air		
PM-10	7.77E-02	G	Output	Airborne	air		
Propionaldehyde	1.69E-05	G	Output	Airborne	air		
Pyrene	4.90E-08	G	Output	Airborne	air		
Selenium	6.25E-05	G	Output	Airborne	air		
Styrene	1.11E-06	G	Output	Airborne	air		
Sulfur dioxide	3.35E+00	G	Output	Airborne	air		
Tetrachloroethylene	1.91E-06	G	Output	Airborne	air		
TOCs, remaining unspeciated	7.66E-03	G	Output	Airborne	air		
Toluene	5.56E-05	G	Output	Airborne	air		
Vanadium	2.22E-04	G	Output	Airborne	air		
Vinyl acetate	3.37E-07	G	Output	Airborne	air		
Xylene (mixed isomers)	1.64E-06	G	Output	Airborne	air		
Zinc (elemental)	2.00E-04	G	Output	Airborne	air		
WATER RELEASES							
Sulfate ion (-4)	3.39E-02	G	Output	Waterborne	surface water		
Suspended solids	8.82E-04	G	Output	Waterborne	surface water		
WASTE							
Coal waste	2.52E+01	G	Output	Solid waste	landfill		
Dust/sludge	9.73E+00	G	Output	Solid waste	landfill		
Fly/bottom ash	6.30E+00	G	Output	Solid waste	landfill		
Low-level radioactive waste	4.29E-03	G	Output	Radioactive waste	landfill		
Uranium, depleted	1.29E-03	G	Output	Radioactive waste	landfill		
RADIOACTIVE AIR EMISSIONS							
Argon-41 (isotope)	3.89E+01	Bq	Output	Radioactivity	air		
Bromine-89 (isotope)	4.50E-06	Bq	Output	Radioactivity	air		

Table 3b. Japanese Electricity Generation Inventory [inputs and outputs per kWh (3.6 MJ)]

Material	Quantity	Units	Input/output	Input/output type	Disposition		
Bromine-90 (isotope)	1.83E-06	Bq	Output	Radioactivity	air		
Cesium-134 (isotope)	1.24E-04	Bq	Output	Radioactivity	air		
Cesium-137 (isotope)	9.33E-04	Bq	Output	Radioactivity	air		
Chromium-51 (isotope)	2.44E-03	Bq	Output	Radioactivity	air		
Cobalt-57 (isotope)	6.57E-06	Bq	Output	Radioactivity	air		
Cobalt-58 (isotope)	8.38E-05	Bq	Output	Radioactivity	air		
Cobalt-60 (isotope)	6.31E-04	Bq	Output	Radioactivity	air		
Iodine-131 (isotope)	2.95E-03	Bq	Output	Radioactivity	air		
Iodine-132 (isotope)	5.99E-04	Bq	Output	Radioactivity	air		
Iodine-133 (isotope)	2.73E+00	Bq	Output	Radioactivity	air		
Iodine-134 (isotope)	3.10E-03	Bq	Output	Radioactivity	air		
Iodine-135 (isotope)	1.56E-04	Bq	Output	Radioactivity	air		
Krypton-85 (isotope)	6.46E+01	Bq	Output	Radioactivity	air		
Krypton-85M (isotope)	3.13E+00	Bq	Output	Radioactivity	air		
Krypton-87 (isotope)	1.17E+00	Bq	Output	Radioactivity	air		
Krypton-88 (isotope)	5.47E+00	Bq	Output	Radioactivity	air		
Manganese-54 (isotope)	3.47E-05	Bq	Output	Radioactivity	air		
Niobium-95 (isotope)	1.38E-06	Bq	Output	Radioactivity	air		
Rubidium-88 (isotope)	1.28E-02	Bq	Output	Radioactivity	air		
Silver-110M (isotope)	4.11E-08	Bq	Output	Radioactivity	air		
Technetium-99M (isotope)	1.85E-07	Bq	Output	Radioactivity	air		
Tritium-3 (isotope)	9.13E+01	Bq	Output	Radioactivity	air		
Xenon-131M (isotope)	5.27E+00	Bq	Output	Radioactivity	air		
Xenon-133 (isotope)	5.05E+01	Bq	Output	Radioactivity	air		
Xenon-133 (isotope)	5.05E+01	Bq	Output	Radioactivity	air		
Xenon-133M (isotope)	7.60E+02	Bq	Output	Radioactivity	air		
Xenon-135 (isotope)	2.87E+01	Bq	Output	Radioactivity	air		
Xenon-135M (isotope)	5.48E-01	Bq	Output	Radioactivity	air		
Xenon-138 (isotope)	1.82E+00	Bq	Output	Radioactivity	air		
Zirconium-95 (isotope)	3.56E-06	Bq	Output	Radioactivity	air		
RADIOACTIVE WATER RELEASES							
Antimony-124 (isotope)	1.92E-02	Bq	Output	Radioactivity	surface water		
Antimony-125 (isotope)	7.67E-02	Bq	Output	Radioactivity	surface water		
Barium-140 (isotope)	1.43E-03	Bq	Output	Radioactivity	surface water		
Cesium-134 (isotope)	5.15E-02	Bq	Output	Radioactivity	surface water		
Cessium-137 (isotope	7.73E-02	Bq	Output	Radioactivity	surface water		
Chromium-51 (isotope)	9.27E-02	Bq	Output	Radioactivity	surface water		
Cobalt-57 (isotope)	2.24E-03	Bq	Output	Radioactivity	surface water		
Cobalt-58 (isotope)	9.13E-01	Bq	Output	Radioactivity	surface water		

Table 3b. Japanese Electricity Generation Inventory [inputs and outputs per kWh (3.6 MJ)]

Material	Quantity	Units	Input/output	Input/output type	Disposition
Cobalt-80 (isotope)	2.40E-01	Bq	Output	Radioactivity	surface water
Iodine-131 (isotope)	4.28E-02	Bq	Output	Radioactivity	surface water
Iodine-132 (isotope)	1.62E-02	Bq	Output	Radioactivity	surface water
Iodine-133 (isotope)	1.83E-02	Bq	Output	Radioactivity	surface water
Iodine-135 (isotope)	1.31E-02	Bq	Output	Radioactivity	surface water
Iron-55 (isotope)	2.18E-01	Bq	Output	Radioactivity	surface water
Iron-59 (isotope)	1.12E-02	Bq	Output	Radioactivity	surface water
Krypton-85M (isotope)	5.77E-02	Bq	Output	Radioactivity	surface water
Lanthanum-140 (isotope)	1.53E-03	Bq	Output	Radioactivity	surface water
Manganese-54 (isotope)	6.11E-02	Bq	Output	Radioactivity	surface water
Molybdenum-99 (isotope)	1.15E+05	Bq	Output	Radioactivity	surface water
Niobium-95 (isotope)	1.57E-02	Bq	Output	Radioactivity	surface water
Ruthenium-103 (isotope)	1.92E-03	Bq	Output	Radioactivity	surface water
Silver-110M (isotope)	2.24E-02	Bq	Output	Radioactivity	surface water
Sodium-24 (isotope)	3.42E-03	Bq	Output	Radioactivity	surface water
Strontium-89 (isotope)	3.70E-03	Bq	Output	Radioactivity	surface water
Strontium-90 (isotope)	8.69E-04	Bq	Output	Radioactivity	surface water
Strontium-95 (isotope)	9.57E-03	Bq	Output	Radioactivity	surface water
Sulfur-136 (isotope)	2.06E-03	Bq	Output	Radioactivity	surface water
Technetium-99M (isotope)	1.34E-03	Bq	Output	Radioactivity	surface water
Tin-113 (isotope)	2.12E-03	Bq	Output	Radioactivity	surface water
Tritium-3 (isotope)	6.83E+02	Bq	Output	Radioactivity	surface water
Xenon-131M (isotope)	7.02E-01	Bq	Output	Radioactivity	surface water
Xenon-133M (isotope)	8.84E-01	Bq	Output	Radioactivity	surface water
Xenon-135 (isotope)	8.05E-01	Bq	Output	Radioactivity	surface water
Zinc-85 (isotope)	1.03E-03	Bq	Output	Radioactivity	surface water

Table 3b. Japanese Electricity Generation Inventory [inputs and outputs per kWh (3.6 MJ)]
3. METHODOLOGY

The U.S.-wide inventory was developed by first compiling inventory data for each of the major generation (fuel-specific) categories used to produce electricity in the U.S., and then creating the U.S.-wide inventory from the fuel-specific inventories. The creation of the U.S.-wide data from the fuel-specific inventories required two particular sets of information: 1997 fuel consumption data and 1997 fuel-specific net electricity generation data (see Table 4). In the majority of cases, one or more pieces of information from these two data sets was needed to convert each input or output into the units of grams per kWh (excluding the radionuclides which were converted to Becquerels per kWh).

Fuel ^a	1997 Fuel Consumption	Units	1997 Net Electricity Generation	1997 Generation % Breakdown
Coal	900,361,000	short tons/yr	1,787,806,000,000	57.23%
Gas	2,968,453	million ft3/yr	283,625,000,000	9.07%
Petroleum	5,256,132	thousand gal/yr	77,753,000,000	2.53%
Nuclear	48,700,000	lbs U ₃ O ₈ /yr	628,644,000,000	20.14%
Hydro ^b			337,233,000,000	10.79%
Renewables ^b			7,462,000,000	0.24%
Total			3,122,522,000,000	100%

Table 4. 1997 U.S. Electricity Utility Summary Statistics

Source: EIA 1999a.

^a This breakdown exludes nonutility electricity genration (non and Independent Power Producers (NPPs or IPPs), which typically contribute about 11% of the U.S. total (EIA 1999).

^b Hydro and renewables were exluded from the calculation of inputs and outputs.

Data on the inputs and outputs for electricity generation were obtained from available sources; when multiple sources of the same type of data were found, those data believed to have the highest quality were utilized (Section 5 addresses data quality). Most data obtained were fuel-specific, however some of the data found were already aggregated to the U.S.-wide level, and thus did not need converting from the fuel-specific values. Thus, for some input/output categories, few calculations were necessary; for others, more complex equations were required to calculate the final input or output data.

As stated earlier in this TM, the final electric grid inventory data (in units of grams/net kWh, for example) will be multiplied by energy use values throughout the life cycle (in units of point-of-use kWhs/functional unit). The two kWhs referred to are different, with the difference being net generated kWhs versus point-of-use kWhs. Due to losses that are associated with moving electrical energy from a point of generation to a point of use, known as 'transmission and distribution (TD) losses,' these must be accounted for in the calculations. It was found in the research for this TM that for 1997 the nationwide TD losses were on the order of 8% of net generation (EIA 1999b). Therefore, to make the kWhs equivalent, the net generation was divided by a TD factor of 1.08 to effectively convert the net kWhs to point-of-use kWhs. The net kWhs are divided by the 1.08 TD factor in all the equations shown in this TM.

APPENDIX E

The following subsections provide the methodology used and basic equations utilized to create the U.S.-wide inventory from the fuel-specific input and output values.

3.1 Inputs

3.1.1 Primary Materials

Primary materials are typically considered in LCA to be those materials that become part of the final product of the process being modeled. For the process of electricity generation, the fuels used to produce the electrical energy are usually considered to be the only 'primary' materials. Thus, for the U.S. and Japanese electric grid inventory data, the primary materials include coal, gas, petroleum and uranium. The U.S. nationwide total quantities of these fuels consumed by utilities in 1997 was obtained from the Energy Information Administration (EIA 1999a for coal, gas and petroleum; EIA 1998b for uranium). The Japanese data were obtained from the EIA's Country Energy Data Report (EIA 1997). Note that the primary material for nuclear-based electricity generation is referred to as "yellowcake," which is the most common physical state on which uranium consumption is based. Yellowcake is a yellowish-brown powder that is the product of the initial milling process that follows mining. The input data were provided on a nationwide basis, thus the equation needed to calculate each fuel's use rate in grams per kWh is shown below:

$$UR_{fuel i} = \frac{Con_{fuel i}}{\frac{Gen_{net}}{TD}} \qquad Eq. 1$$

where,

$UR_{fuel i}$	=	nationwide input use rate per unit of electricity for fuel <i>i</i> , where the four
		fuels are coal, gas, petroleum and uranium (grams/kWh),
$\operatorname{Con}_{\operatorname{fuel} i}$	=	annual consumption rate for fuel <i>i</i> (mass or volume/yr; converted to grams
		using density as necessary, see Attachment E, Table E5),
Gen _{net}	=	net annual nationwide electricity generation (kWh/yr), and
TD	=	nationwide average transmission and distribution losses factor (percent).

In calculating the primary material consumption rates for each fuel, material densities and conversion factors for mass, volume and energy were utilized where needed, depending on whether the units of consumption were in mass or volume (see Attachment E, Table E5).

3.1.2 Ancillary Materials

Ancillary materials are considered to be those materials that help the process function or work, yet do not become part of the final product. In generating electricity, all materials used, except the fuel itself, are typically considered to be ancillary materials. The ancillary materials accounted for in the U.S. electric grid inventory data are limestone and lime which are sulfur dioxide removal system catalysts typically used in coal firing, and cooling water which is consumed in generating electricity from all fuel types.

The limestone and lime values were derived by utilizing data from approximately five different sources, primarily the EIA, the Acid Rain database and direct contact with utility employees. In short, individual lime and limestone annual consumption rates were developed in units of pounds per year for each plant in the U.S. that utilizes a lime- or limestone-based flue gas desulfurization (FGD) system, and then those values were added to obtain the total poundage or tonnage consumed annually in the U.S. (tons per year). At that point, Equation 1 could be used (calculating only for coal) to obtain the quantity of lime or limestone consumed in grams per kWh generated. Cooling water consumption in units of gallons per kWh was obtained from the California Energy Commission (CEC 1979; cited in Paul Gipe's *Wind Energy Comes of Age*, John Wiley & Sons, 1995). Since cooling water is consumed during electricity generation for each of the fuel types followed in this inventory data set, one equation was used to calculate the nationwide average cooling water requirements:

$$UR_{water} = \frac{\sum_{i=1}^{4} (Con_{water} \times Gen)_{fuel i}}{\frac{Gen_{net}}{TD}} Eq. 2$$

where,

,		
Ur _{water}	=	nationwide input use rate per unit of electricity for water (grams/kWh),
Con _{water}	=	average consumption rate per unit of electricity for water for fuel <i>i</i> , where
		the four fuels are coal, gas, petroleum and uranium (gallons/kWh;
		converted to grams using density, see Attachment E, Table E5),
Gen	=	net annual electricity generation rate for fuel <i>i</i> (kWh/yr),
Gen _{net}	=	net annual nationwide electricity generation (kWh/yr), and
TD	=	nationwide average transmission and distribution losses factor (percent).

3.2 Outputs

3.2.1 Air Emissions

Several sources of air emissions data where evaluated in compiling the electric grid inventory data. The evaluation revealed that AP-42 (EPA 1996) data are the most complete source of speciated air emissions data *that are easily accessible and do not require a substantial investment to obtain*. Thus, AP-42 data were used as the foundation for the air emissions estimates. ('AP-42' is the EPA's emission factors data set that addresses the type and quantity of air pollutants that result from over 200 major industries, point sources and mobile sources.)

The evaluation also revealed a few sources of higher quality data for some pollutants, and in those cases, that information was used to either augment or replace the AP-42 factors. Specifically, the air pollutant release rates for criteria pollutants were obtained from the EPA (1998b) on a nationwide annual basis and used instead of the AP-42 factors for those pollutants (the values covered all fossil fuel-based generation categories and were in units of pounds per year). The criteria pollutants include carbon dioxide (CO_2), sulfur dioxide (SO_2), nitrogen oxides

 (NO_x) , carbon monoxide (CO), lead (Pb) and particulate matter ten micrometers or less in diameter (PM-10), and the following equation was used to obtain the emission rates for these pollutants in units of grams/kWh:

$$RR_{air-criteria i} = \frac{Rel_{air-criteria i}}{\frac{Gen_{net}}{TD}} Eq. 3$$

where,

RR _{air-criteria i}	=	nationwide release rate per unit of electricity for criteria pollutant <i>i</i>
		(grams/kWh),
Rel _{air-criteria i}	=	annual release rate for criteria pollutant <i>i</i> (lbs/yr),
Gen _{net}	=	net annual nationwide electricity generation (kWh/yr), and
TD	=	nationwide average transmission and distribution losses factor
		(percent).

Nuclear power plants produce electricity without combusting fuel, thus no criteria pollutants are emitted to the local air environment during nuclear power production. However, nuclear-based electricity generation does produce some airborne radionuclides, which are addressed in Section 3.2.4. No airborne releases from nuclear-based generation are addressed here.

The air pollutants accounted for in this inventory calculated from AP-42 air emission factors are shown in Table 5. Of these air emission factors, Table 5 shows the number of pollutants in each category, what fuels AP-42 has release data for, and whether the AP-42 data reflects controlled or uncontrolled release of those pollutants.

The equation used to calculate the nationwide emissions of these non-criteria pollutants is Equation 4. The equation shows a summation of three release and consumption rates, which was adjusted for the number of fuels from which the pollutants were listed in AP-42 (shown in Table 5).

For all of the pollutants in Table 5, AP-42 lists only one emission factor per pollutant per fuel (for example, pounds of benzene released per ton of coal burned). Thus, each factor represents a combined emissions estimate for the various technologies used to fire each fuel.

In augmenting the AP-42 data, it was also determined that the nationwide methane (CH_4) air releases generated during coal mining (EPA 1998c) are significant, and were added to the emissions estimates of CH_4 from coal and petroleum combustion. These emissions were presented in units of nationwide cubic feet of methane released per year.

Pollutant Category/Pollutant	# of factors provided	Fuels from I	Fuels from Which Pollutants Were Listed in AP-42		
		Coal	Gas	Petroleum	
Speciated organic compounds	37	v	~	~	Controlled
Trace metals	13	~	~	~	Controlled
Polycryclic aromatic hydrocarbons	16	~			Controlled
Dioxins & furans	16	~			Controlled
Methane	1	~		~	Uncontrolled
Nitrous oxide	1	~	~	~	Uncontrolled
Hydrogen chloride	1	~			Uncontrolled
Hydrogen fluoride	1	~			Uncontrolled
Total organic compounds	1	v	~	 ✓ 	Uncontrolled
Total nonmethane organic compounds	1	~		v	Uncontrolled

Table 5. Air Pollutant Information from AP-42

Source: EPA 1996.

$$RR_{air-other j} = \frac{\sum_{i=1}^{3} \left(Rel_{air-other j} \times Con \right)_{fuel i}}{\frac{Gen_{net}}{TD}} \qquad Eq. 4$$

where,

which c,		
$\operatorname{Rr}_{\operatorname{air-other} i}$	=	nationwide release rate for each pollutant <i>j</i> (grams/kWh),
Rel _{air-other i}	=	release rate per unit of fuel for each air pollutant <i>j</i> from fossil fuel <i>i</i> (coal -
2		lbs/ton, gas - lbs/million cubic feet, petroleum - lbs/thousand gallons),
Con	=	annual consumption rate of fuel <i>i</i> (coal - tons/yr, gas - million cubic
		feet/yr, petroleum - thousand gallons/yr),
Gen _{net}	=	net annual nationwide electricity generation (kWh/yr), and
TD	=	nationwide average transmission and distribution losses factor (percent).

3.2.2 Solid Wastes

For the U.S. electric grid inventory data, two types of solid waste exist: coal-fired generation nonhazardous solid wastes and nuclear-based generation radioactive solid wastes. [The term 'solid waste' as it is used in this report applies to the Resource Conservation and Recovery Act (RCRA) definition which is defined in the U.S. Federal Code of Regulations (40 CFR 261).] For coal-fired generation, the solid wastes values were obtained from a 1994 Oak Ridge National Laboratory (ORNL) report entitled "Estimating Externalities of Coal Fuel Cycles," and were provided in units of tons of waste produced per gigawatthour (GWh) generated. The calculation needed to obtain the nationwide releases is as follows:

<i>RR</i> =	_ <i>Rel</i>	$solid_i \times Gen_{coal}$	
solid i		Gen _{net} Eq. 5	
		TD	
where,			
RR _{solid i}	=	nationwide release rate per unit of electricity for solid waste <i>i</i> from	coal
		fired generation (grams/kWh),	
Rel _{solid i}	=	release rate per unit of electricity for solid waste <i>i</i> from coal-fired	
		generation (tons/GWh),	
Gen _{coal}	=	net annual electricity generation rate for coal (MWh/yr),	
Gen _{net}	=	net annual nationwide electricity generation (kWh/yr), and	
TD	=	nationwide average transmission and distribution losses factor (perc	cent)

For the two nuclear-based generation wastes quantified, spent fuel and low-level radioactive waste (LLRW), two data sources were utilized. The spent fuel data information was obtained from the EIA in units of pounds per year and the LLRW data from an expert in the nuclear electricity generation industry (Loiselle 1998) in units of cubic feet per year (see Attachment E, Table E5 for LLRW conversion factor). The calculation used to derive the quantity of both nuclear wastes generated in units of grams per kWh was similar to Equations 1 and 3, and is shown below:

$$GR_{nuclear i} = \frac{Gen_{nuclear i}}{\frac{Gen_{net}}{TD}} Eq. 6$$

where,

$GR_{nuclear i}$	=	nationwide generation rate per unit of electricity for nuclear waste <i>i</i>
		(grams/kWh),
Gen _{nuclear i}	=	nationwide annual generation rate for nuclear waste <i>i</i> (spent fuel - lbs/yr,
		LLRW - ft ³ /yr; converted to grams using density as necessary, see
		Attachment E, Table E5),
Gen _{net}	=	net annual nationwide electricity generation (kWh/yr), and
TD	=	nationwide average transmission and distribution losses factor (percent).

3.2.3 Water Releases

Water release information was obtained on coal-fired generation. Radioactive water release information is presented in Section 3.2.4. Information on only three water pollutants was obtained (ORNL 1994). Data were in units of tons/GWh, and thus used an equation like Equation 5 to calculate the nationwide release per unit of electricity generated:

$$RR_{water i} = \frac{Rel_{water i} \times Gen_{coal}}{\frac{Gen_{net}}{TD}}$$
 Eq. 7 where,

RR _{water i}	=	nationwide release rate per unit of electricity for water pollutant <i>i</i> from
		coal-fired generation (grams/kWh),
Rel _{water i}	=	release rate per unit of electricity for water pollutant <i>i</i> from coal-fired
		generation (tons/GWh),
Gen _{coal}	=	net annual electricity generation rate for coal (MWh/yr),
Gen _{net}	=	net annual nationwide electricity generation (kWh/yr), and
TD	=	nationwide average transmission and distribution losses factor (percent)

3.2.4 Radionuclides

During the operation of nuclear electricity generation facilities, both airborne and waterborne radioactive releases are generated that are not directly related to a facility's capacity to generate power (kWs) or to the quantity of power it generates over time (kWhs). Data were obtained on the radioactive releases of many nuclear facilities in the U.S. from an ORNL report which addressed the externalities of nuclear fuel cycles (ORNL 1995). This report presented averaged release rate information for 31 airborne and 35 waterborne radioactive releases, which were from direct measurements at many U.S.-based nuclear facilities, all of which were presented in units of Curies per year. Although the releases where found not to relate directly to power generation or power generating capacity, some scale was needed to convert the quantity of release from multiple facilities to a quantity of release for all the facilities in use in the U.S. Therefore, due to a lack of any other identifiable scaling mechanism, the quantity of power generated was utilized. The Curies per year value for each radioactive release was first converted into Curies per MWh by dividing by the number of MWhs generated by the average facility identified in the report. Then the values were converted into units of Becquerels per kWh. The following equation was utilized for converting airborne and waterborne radionuclides to the nationwide grid:

$$RR_{radionuclide i} = \frac{Rel_{radionuclide i} \times Gen_{nuclear}}{\frac{Gen_{net}}{TD}} Eq. 8$$

where,

RR _{radionuclide i}	=	nationwide release rate per unit of electricity for radionuclide <i>i</i>
		(Becquerels/kWh),
Rel _{radionuclide i}	=	averaged release rate per unit of electricity for radionuclide <i>i</i>
		(Curies/MWh),
Gen _{nuclear}	=	annual electricity generation rate for nuclear (MWh/yr),
Gen _{net}	=	net annual nationwide electricity generation (kWh/yr), and
TD	=	nationwide average transmission and distribution losses factor (percent)

4. FUEL-SPECIFIC RESULTS

Tables 6 through 9 present the inputs and outputs from each of the individual fuel inventories. The data in those tables were aggregated into the U.S.-wide average data for electricity generation (Table 3a) and Japanese average data (Table 3b). Additionally, the spreadsheets from which Tables 6 through 9 were derived are shown in Attachments A through E and provide some additional information about each inventory.

Material/pollutant	Quantity	Unit	Material/pollutant	Quantity	Unit
INPUTS			OUTPUTS	(continued)	
Primary Materials			Air Emissions (continued)		
Coal	9.00E+08	tons/yr	TRACE METALS (continued	d)	
			Beryllium	2.10E-05	lbs/t
Ancillary Materials			Cadmium	5.10E-05	lbs/t
Limestone	1.21E+07	tons/yr	Chromium	2.60E-04	lbs/t
Lime	5.31E+06	tons/yr	Chromium (VI)	7.90E-05	lbs/t
Cooling water	4.90E-01	gal/kWh	Cobalt	1.00E-04	lbs/t
			Magnesium	1.10E-02	lbs/t
OUTPUTS	5		Manganese	4.90E-04	lbs/te
Air Emissions			Mercury	8.30E-05	lbs/te
Methane	4.00E-02	lbs/ton	Nickel	2.80E-04	lbs/te
Nitrous oxide	3.00E-02	lbs/ton	Selenium	1.30E-03	lbs/to
Hydrogen chloride	1.20E+00	lbs/ton	POLYCYCLIC AROMATIC	HYDROCARBON	IS (Pz
Hydrogen fluoride	1.50E-01	lbs/ton	Biphenyl	1.70E-06	lbs/to
TOC	3.00E-01	lbs/ton	Acenaphthene	5.10E-07	lbs/te
TNMOC	6.00E-02	lbs/ton	Acenaphthylene	2.50E-07	lbs/te
SPECIATED ORGANIC COMP	OUNDS		Anthracene	2.10E-07	lbs/te
Acetaldehyde	5.70E-04	lbs/ton	Benzo(a)anthracene	8.00E-08	lbs/te
Acetophenone	1.50E-05	lbs/ton	Benzo(a)pyrene	3.80E-08	lbs/te
Acrolein	2.90E-04	lbs/ton	Benzo(b,j,k)fluoranthene	1.10E-07	lbs/to
Benzene	1.30E-03	lbs/ton	Benzo(g,h,i)perylene	2.70E-08	lbs/to
Benzyl chloride	7.00E-04	lbs/ton	Chrysene	1.00E-07	lbs/to
Bis(2-ethylhexyl)phthalate (DEHP)	7.30E-05	lbs/ton	Fluoranthene	7.10E-07	lbs/to
Bromoform	3.90E-05	lbs/ton	Fluorene	9.10E-07	lbs/to
Carbon disulfide	1.30E-04	lbs/ton	Indeno(1,2,3-cd)pyrene	6.10E-08	lbs/to
2-Chloroacetophenone	7.00E-06	lbs/ton	Naphthalene	1.30E-05	lbs/to
Chlorobenzene	2.20E-05	lbs/ton	Phenanthrene	2.70E-06	lbs/te
Chloroform	5.90E-05	lbs/ton	Pyrene	3.30E-07	lbs/to
Cumene	5.30E-06	lbs/ton	5-Methyl chrysene	2.20E-08	lbs/to
Cyanide	2.50E-03	lbs/ton	DIOXINS & FURANS		
2,4-Dinitrotoluene	2.80E-07	lbs/ton	2,3,7,8-TCDD	1.43E-11	lbs/to
Dimethyl sulfate	4.80E-05	lbs/ton	Total TCDD	9.28E-11	lbs/to
Ethyl benzene	9.40E-05	lbs/ton	Total PeCDD	4.47E-11	lbs/to

 Table 6. Inputs and Outputs for Coal-Fired Electricity Generation in the United States^a

Material/pollutant	Quantity	Unit	Material/pollutant	Quantity	Unit
Ethyl chloride	4.20E-05	lbs/ton	Total HxCDD	2.87E-11	lbs/ton
Ethylene dichloride	4.00E-05	lbs/ton	Total HpCDD	8.34E-11	lbs/ton
Ethylene dibromide	1.20E-06	lbs/ton	Total OCDD	4.16E-10	lbs/ton
Formaldehyde	2.40E-04	lbs/ton	Total PCDDd	6.66E-10	lbs/ton
Hexane	6.70E-05	lbs/ton	2,3,7,8-TCDF	5.10E-11	lbs/ton
Isophorone	5.80E-04	lbs/ton	Total TCDF	4.04E-10	lbs/ton
Methyl bromide	1.60E-04	lbs/ton	Total PeCDF	3.53E-10	lbs/ton
Methyl chloride	5.30E-04	lbs/ton	Total HxCDF	1.92E-10	lbs/ton
Methyl ethyl ketone	3.90E-04	lbs/ton	Total HpCDF	7.68E-11	lbs/ton
Methyl hydrazine	1.70E-04	lbs/ton	Total OCDF	6.63E-11	lbs/ton
Methyl methacrylate	2.00E-05	lbs/ton	Total PCDFd	1.09E-09	lbs/ton
Methyl tert butyl ether	3.50E-05	lbs/ton	COAL MINE METHANE EMISSI	ONS	
Methylene chloride	2.90E-04	lbs/ton	Methane	1.52E+11	ft ³ /yr
Phenol	1.60E-05	lbs/ton			
Propionaldehyde	3.80E-04	lbs/ton	Solid Wastes		
Tetrachloroethylene	4.30E-05	lbs/ton	Dust/sludge	5.51E+01	tons/GWh
Toluene	2.40E-04	lbs/ton	Coal waste	1.43E+02	tons/GWh
1,1,1-Trichloroethane	2.00E-05	lbs/ton	Fly/bottom ash	3.57E+01	tons/GWh
Styrene	2.50E-05	lbs/ton			
Xylenes	3.70E-05	lbs/ton	Water Releases		
Vinyl acetate	7.60E-06	lbs/ton	Dissolver	2.78E-01	tons/GWh
TRACE METALS			Suspended solids	5.00E-03	tons/GWh
Antimony	1.80E-05	lbs/ton	Sulfate	1.92E-01	tons/GWh

Table 6. Inputs and Outputs for Coal-Fired Electricity Generation in the United States^a

Material/pollutant	Quantity	Unit	Pollutant	Quantity	Unit	
INPUTS			OUTPUTS (continued)			
Primary Material			Air Emissions (continue	ed)		
Gas	3.00E+12	yr 3/yr	SPECIATED ORGANIC	COMPOUNDS (co	ntinued)	
			Naphthalene	2.40E-04	lbs/Mft3	
Ancillary Material			Phenanthrene	1.00E-05	lbs/Mft3	
Water	2.50E-01	gal/kWh	Pyrene	5.01E-06	lbs/Mft3	
			Toluene	2.20E-03	lbs/Mft3	
OUT	PUTS		TRACE METALS			
Air Emissions			Arsenic	2.30E-04	lbs/Mft3	
Nitrous oxide	2.20E+00	lbs/Mft3	Barium	2.40E-03	lbs/Mft3	
Filterable PM	1.50E+00	lbs/Mft3	Chromium	1.10E-03	lbs/Mft3	
Condensible PM	1.50E+00	lbs/Mft3	Cobalt	1.20E-04	lbs/Mft3	
TOC	1.70E+00	lbs/Mft3	Copper	2.51E-04	lbs/Mft3	
SPECIATED ORGANIC (COMPOUNDS		Manganese	3.81E-04	lbs/Mft3	
Fluoranthene	3.01E-06	lbs/Mft3	Molybdenum	5.81E-04	lbs/Mft3	
Formaldehyde	1.55E-01	lbs/Mft3	Nickel	3.61E-03	lbs/Mft3	
2-Methylnaphthalene	9.02E-06	lbs/Mft3	Vanadium	3.21E-03	lbs/Mft3	

Table 7. Inputs and Outputs for Gas-Fired Electricity Generation in the United States

Material/Pollutant	Quantity	Unit	Pollutant	Quantity Unit		
INPUTS			OUTPUTS (continued)			
Primary Material			Air Emissions (continued			
Petroleum			SPECIATED ORGANIC CC	OMPOUNDS (continued)		
			OCDD	3.10E-09 lbs/kgal		
Ancillary Material			Phenanthrene	1.05E-05 lbs/kgal		
Water	4.30E-01	gal/kWh	Pyrene	4.25E-06 lbs/kgal		
			Toluene	6.20E-03 lbs/kgal		
OUTPU	JTS		1,1,1-Trichloroethane	2.36E-04 lbs/kgal		
Air Emissions			o-xylene	1.09E-04 lbs/kgal		
Methane	2.66E-01	l lbs/kgal	TRACE METALS			
Nitrous oxide	1.10E-01	l lbs/kgal	Antimony	5.25E-03 lbs/kgal		
TOC	9.93E-01	l lbs/kgal	Arsenic	1.28E-03 lbs/kgal		
TNMOC	7.26E-01	l lbs/kgal	Barium	2.57E-03 lbs/kgal		
SPECIATED ORGANIC CO	MPOUNDS		Beryllium	4.71E-05 lbs/kgal		
Acenaphthene	2.11E-05	5 lbs/kgal	Cadmium	4.67E-04 lbs/kgal		
Acenaphthylene	2.53E-07	7 lbs/kgal	Chloride	3.47E-01 lbs/kgal		
Anthracene	1.22E-06	6 lbs/kgal	Chromium	1.28E-03 lbs/kgal		
Benzene	2.14E-04	lbs/kgal	Chromium (VI)	2.48E-04 lbs/kgal		
Benz(a)anthracene	4.01E-06	6 lbs/kgal	Cobalt	6.02E-03 lbs/kgal		
Benzo(b,k)fluoranthene	1.48E-06	6 lbs/kgal	Copper	1.76E-03 lbs/kgal		
Benzo(g,h,i)perylene	2.26E-06	6 lbs/kgal	Fluoride	3.73E-02 lbs/kgal		

Table 8. Inputs and Outputs for Petroleum-Fired Electricity Generation in the U.S.

Material/Pollutant	Quantity	Unit		Pollutant	Quantity	Unit
Chrysene	2.38E-06	lbs/kgal		Manganese	2.94E-03	lbs/kgal
Dibenzo(a,h)anthracene	1.67E-06	lbs/kgal]	Mercury	1.31E-04	lbs/kgal
Ethylbenzene	6.36E-05	lbs/kgal		Molybdenum	7.87E-04	lbs/kgal
Fluoranthene	4.84E-06	lbs/kgal		Nickel	8.09E-02	lbs/kgal
Fluorene	4.47E-06	lbs/kgal		Phosphorous	9.46E-03	lbs/kgal
Formaldehyde	3.30E-02	lbs/kgal		Selenium	6.83E-04	lbs/kgal
Indo(1,2,3-cd)pyrene	2.14E-06	lbs/kgal]	Vanadium	3.18E-02	lbs/kgal
Naphthalene	1.13E-03	lbs/kgal	1	Zinc	2.91E-02	lbs/kgal

Table 9. Inputs and Outputs for Nuclear-Based Electricity Generation in the U.S.

Material/pollutant	Quantity	Unit	Pollutant	Quantity	Unit	
INPUTS		OUTPUTS (continued)				
Primary Material			Waterborne Radionuclide Emissions (continued)			
Uranium oxide	4.87E+07	lbs/yr	Xe-135M	4.40E-08	Curies/MWh	
("yellowcake")						
			Cs-137	7.49E-11	Curies/MWh	
Ancillary Material	•		Xe-138	1.46E-07	Curies/MWh	
Water	6.20E-01	gal/kWh	Waterborne Radionuclide En	nissions		
			T-3	5.48E-05	Curies/MWh	
OUTI	PUTS		Na-24	2.75E-10	Curies/MWh	
Solid Wastes			Cr-51	7.44E-09	Curies/MWh	
Spent fuel	5.29E+06	lbs/yr	Mn-54	4.90E-09	Curies/MWh	
LLRW	2.21E+05	ft³/yr	Fe-55	1.75E-08	Curies/MWh	
			Co-57	1.80E-10	Curies/MWh	
Airborne Radionuclide E	missions		Co-58	7.33E-08	Curies/MWh	
T-3	7.33E-06	Curies/MWh	Fe-59	9.00E-10	Curies/MWh	
Ar-41	3.13E-06	Curies/MWh	Co-80	1.92E-08	Curies/MWh	
Cr-51	1.96E-10	Curies/MWh	Zn-85	8.27E-11	Curies/MWh	
Mn-54	2.78E-12	Curies/MWh	Kr-85M	4.63E-09	Curies/MWh	
Co-57	5.27E-13	Curies/MWh	Sr-89	2.97E-10	Curies/MWh	
Co-58	6.73E-12	Curies/MWh	Sr-90	6.97E-11	Curies/MWh	
Co-60	5.06E-11	Curies/MWh	Nb-95	1.26E-09	Curies/MWh	
Kr-85	5.18E-06	Curies/MWh	Sr-95	7.68E-10	Curies/MWh	
Kr-85M	2.51E-07	Curies/MWh	Mo-99	9.25E-03	Curies/MWh	
Kr-87	9.35E-08	Curies/MWh	Tc-99M	1.08E-10	Curies/MWh	
Rb-88	1.03E-09	Curies/MWh	Ru-103	1.54E-10	Curies/MWh	
Kr-88	4.39E-07	Curies/MWh	Ag-110M	1.80E-09	Curies/MWh	
Br-89	3.62E-13	Curies/MWh	Sn-113	1.70E-10	Curies/MWh	
Br-90	1.47E-13	Curies/MWh	Sb-124	1.54E-09	Curies/MWh	
Nb-95	1.11E-13	Curies/MWh	Sb-125	6.15E-09	Curies/MWh	
Zr-95	2.86E-13	Curies/MWh	I-131	3.43E-09	Curies/MWh	
Tc-99M	1.48E-14	Curies/MWh	Xe-131M	5.64E-08	Curies/MWh	

Material/pollutant	Quantity	Unit	Pollutant	Quantity	Unit
Ag-110M	3.30E-15	Curies/MWh	I-132	1.30E-09	Curies/MWh
I-131	2.37E-10	Curies/MWh	Xe-133	8.66E-06	Curies/MWh
Xe-131M	4.23E-07	Curies/MWh	I-133	1.47E-09	Curies/MWh
I-132	4.80E-11	Curies/MWh	Xe-133M	7.10E-08	Curies/MWh
Xe-133	6.10E-05	Curies/MWh	Cs-134	4.13E-09	Curies/MWh
I-133	2.19E-07	Curies/MWh	I-135	1.06E-09	Curies/MWh
Xe-133M	4.06E-06	Curies/MWh	Xe-135	6.46E-08	Curies/MWh
Cs-134	9.93E-12	Curies/MWh	s-136	1.65E-10	Curies/MWh
I-134	2.49E-10	Curies/MWh	Cs-137	6.20E-09	Curies/MWh
Xe-135	2.30E-06	Curies/MWh	Ba-140	1.14E-10	Curies/MWh
I-135	1.25E-11	Curies/MWh	La-140	1.23E-10	Curies/MWh

Table 10 presents the number of inventory data points for each fuel type by input or output category. As can easily be seen, most of the fuel-specific inventories are dominated by air pollution data. Of the main categories of air pollutants considered, the following breakdown lists those emissions that are the biggest contributors (have the largest emission rates in overall quantity) to each part of each fuel inventory:

- Coal Cyanide (speciated organic compounds), magnesium (metal), naphthalene [polycyclic aromatic hydrocarbons (PAH)] and the total octochlorodibenzo-p-dioxins (dioxins and furans)
- Gas Formaldehyde (speciated organic compound), nickel (metal)
- Petroleum Formaldehyde (speciated organic compound), chloride (metal)
- Nuclear Xenon-133 (airborne radionuclide), molybdenum-99 (waterborne radionuclide)

Note that in Tables 6 through 9, the units for the inputs and outputs were left as originally found in their source. All fuel-specific inputs and outputs (excluding radioactive releases) were converted to the units desired for the U.S.-wide electric grid of grams/kWh during aggregation into that inventory data set.

Tuble 10. Mulliber of Inputs & Outputs Within Luch Inventory							
Inputs & Outputs	Coal	Gas	Petroleum	Nuclear			
Primary materials	1	1	1	1			
Ancillary materials	3	1	1	1			
Air releases	89	29	51	31			
Water releases	3			35			
Solid & hazardous wastes	3			2			
Total	99	31	53	70			

Table 10. Number of Inputs & Outputs Within Each Inventory

5. DATA SOURCES & QUALITY

Source and quality information for the data presented in this TM are detailed in Table 11. In general, data assigned higher quality ratings were directly measured and represent 1997 data. As data required more calculation or estimation, or were from a previous year, data quality was reduced. Additional comments about data source and quality are in the following subsections.

5.1 Nonfuel specific data

The criteria pollutant air emission values used in this inventory all came from the *National Air Quality & Emissions Trends Report, 1997.* While this report does state that the values supplied are "estimates of the amount and kinds of pollutants being emitted ... based upon best available engineering calculations," (EPA 1998b) the EPA used measured air emission rates where feasible, and thus these data were given 'Average' data quality ratings.

5.2 Coal

As the leading electricity-producing fuel in the U.S., coal has accounted for between 40% and 60% of the kWh produced by utilities in the national grid since the 1930s (NEI 1997). In 1997, coal-fired generation accounted for just over 57% of all utility-generated electricity.

Of the coal inventory data, the coal quantity used annually and the methane generated from coal mining were both directly measured data for 1997 and thus were given 'Excellent' data quality ratings. The data for speciated air emissions that came from AP-42 were deemed 'Poor,' due primarily to the following two facets of the data. First, in averaging the data quality ratings that the EPA applies to their AP-42 factors, an average rating of approximately 3 is calculated, indicating by their own standards an 'average' rating. (EPA's data quality ratings are A, B, C, D and E, where, for example, A, C and E represent 'Excellent,' 'Average' and 'Poor' respectively. This alphabetically based system was temporarily converted into a numerical system of 1 through 5 where 1 corresponds to A and 5 corresponds to E to calculate the required averages.) Second, the bulk of the AP-42 data was dated January 1995; updates were included several times since then (two in 1996 and one in 1998), however, each update included only small changes to the whole AP-42 emission factor data set.

The cooling water use (for all fuels) and the solid waste and water release data for coal were assigned a data quality rating of 'Unknown' due to a lack of information on the original data source.

I/O Type ^a	Data	Data Source/ Reference	Data Source Comments	Data Quality ^b	Data Quality Explanation
NONF	UEL SPECIFIC				
AR	Criteria pollutants	EPA 1998b	Although the EPA uses measured data wherever feasible in calculating the emissions included in this report, much data required estimating to effectively model the national totals for certain pollutant categories.	Average	Although the bulk of the data used in this analysis were measured, some estimates were required to obtain emissions information from particular industries before aggregating to obtain the U.S wide totals.
COAL					
PM	Coal	EIA 1998c	Measured; required for regulatory recording purposes.	Excellent	Measurements provide the highest quality data.
AM	Limestone and lime	EIA 1997	The primary data used in calculating the limestone and lime usage were measured, however approximately six sources of information were accessed to obtain the necessary information to derive final factors.	Average	Of the sources referenced for this calculation, three utilized measured data, while the remaining utilized average or poor quality data. Thus, the overall rating is 'Average.'
АМ	Cooling water	CEC 1979	Only source located in which cooling water requirements were detailed for the main electricity generation categories.	Unknown	CEC did not list its data sources.
SW & WR	All	ORNL 1994	Original source of data in the ORNL 1994 report was Meridian Corporation 1989, however, could not be located.	Unknown	Original data document could not be found.
AR	All noncriteria pollutants	EPA 1996	AP-42 factors have a self-assigned 'Average' to 'Poor' quality rating.	Poor	Given 'Poor' rating for several reasons, including primarily that 1) overall AP-42 self-assigned ratings are 'Average' and 2) data applies to 1995.
AR	Methane from coal mining	EPA 1998c	Measured; reported for regulatory recording purposes.	Excellent	Measurements provide the highest quality data.
GAS &	PETROLEUM				
PM	Gas and petroleum	EIA 1998c	Measured; required for regulatory recording purposes.	Excellent	Measurements provide the highest quality data.
АМ	Cooling water	CEC 1979	Only source located in which cooling water requirements were detailed for the main electricity generation categories.	Unknown	CEC did not list its data sources.
AR	All noncriteria pollutants	EPA 1996	AP-42 factors have a self-assigned 'Poor' quality rating for gas and 'Average' to 'Poor' quality rating for petroleum.	Poor	Given 'Poor' rating for several reasons, including primarily that 1) overall AP-42 self-assigned ratings are 'Average' and 2) data applies to 1995.
URAN	IUM				
РМ	Uranium	EIA 1998b	Measured data, yet had to scale using an unproven scaling mechanism.	Average	Measurements provide the highest quality data, yet the scaling mechanism utilized introduces potential error.
АМ	Cooling water	CEC 1979	Only source located in which cooling water requirements were detailed for the main electricity generation categories.	Unknown	CEC did not list its data sources.
SW	Spent fuel	EIA 1996	Estimates the quantity of spent fuel that will be generated in the US in 1997, using measured historical records and knowledge of what facilities will be changing operating patterns in future years (from 1996 perspective).	Excellent	Although not directly measured, use of recently measured data along with in-depth knowledge of future industry changes to project value earns an 'Excellent' rating.
SW	LLRW	Loiselle 1998	Estimated by an industry expert.	Average	Lack of any measured data gives way to an 'Average' rating.
RR	Airborne radionuclides	ORNL 1995	Averaged measured data from nuclear generation facilities.	Excellent	Measurements provide the highest quality data.
RR	Waterborne radionuclides	ORNL 1995	Averaged measured data from nuclear generation facilities.	Excellent	Measurements provide the highest quality data.

Table 11. Data Sources and Quality Information for the U.S. Electric Grid Inventories

^a Input/Output (I/O) types: PM = primary material, AM = ancillary material, SW = solid waste, AR = airborne release, WR = water release,

RR = radioactive release.

^b The data quality ratings given were assigned to one of the following four data quality categories: Excellent, Average, Poor and Unknown.

5.3 Gas & Petroleum

Each of these data sets utilize the EIA's *Electric Power Annual* for the primary fuel consumption estimates and AP-42 data for the air emissions. The EIA data were given 'Excellent' data quality marks as these data are reported as direct measurements and applicable to 1997. The air emissions were given a quality rating of 'Poor' for the same two reasons the coal inventory data AP-42 estimates were given that rating.

5.4 Uranium

For the nuclear-related inputs and outputs, the largest category of input or output type information was radionuclide emissions. The radionuclide emission information contained in the report was obtained by averaging radionuclide emissions data from 36 different pressurized water reactors (PWRs) across the U.S. (out of just over 110 nuclear reactors total in the U.S.). As discussed previously, the emissions were not found to directly correlate to electricity generation (kWh) or generating capacity (kW), but "were more likely affected by random events within the reactor, such as fuel pin cladding failures, leaks in the primary coolant loop and steam-generator tube leaks" (ORNL 1995). Thus, all factors considered, the overall data quality rating for these emissions was deemed 'Average.' The mass of uranium consumed annually was given the 'Excellent' rating as this information is measured through reporting supplied to the EIA and applicable to 1997.

Of the remaining inputs and outputs, most received higher quality ratings except for the cooling water values which received the 'Unknown' rating due to a lack of information on the original data source.

6. DATA LIMITATIONS

Several limitations of the U.S. electrical energy grid inventory data relate to the exclusion of either entire generation categories (e.g., hydro) or life-cycle substages (see Table 2). For example, no data are readily available for the life-cycle substages of ore and fuel transportation and preliminary ore or fuel processing. Examples of ore and fuel transportation and processing burdens for which insufficient data are available include the emissions generated during the processing of crude oil to produce fuel oils Nos. 6 and 2, and the energy use and wastes generated during the processing of uranium ore into fuel pellets.

With regard to the exclusion of entire generation categories, the renewable and hydroelectric generating categories have not been included in this U.S.-wide inventory. Renewables accounted for only about 0.24% of the total electricity generated in the U.S. in 1997, and are expected to have greatly reduced impacts when compared to the other fuel types. Hydropower was omitted, as stated previously, due to the scarcity of data on hydroelectric inputs and outputs, and also that the CDP LCIA methodology does not account directly for the effects of habitat destruction (expected to be one of the largest impacts from hydroelectric generation once quantified).

Other limitations of the data are related to the use of AP-42 emissions factors, primarily due to the fact that EPA's own ratings of their factors are, at best, 'Average' (see discussion in Section 5.2). Other data limitations include the small number of sources from which much of the

data have been obtained. Not necessarily as much a limitation of the data as a limitation in the sources of the data, this is still an area for quality improvement. Also, as noted previously, nonutility electricity generation in the U.S. is excluded from this inventory, and accounts for about 11% of the total electricity generated annually.

There are some implications that can be derived from the limitations mentioned here on the U.S.-wide electric grid inventory data. These implications include the following:

- Since there are currently no other sources of detailed air emission information like AP-42, there is no way of knowing just how accurate their estimates are (possibly predicting a bias in one direction or the other). It appears that the reason that the EPA's quality ratings of their AP-42 data tend toward 'Average' is that in quite a few cases, emissions test were run on only a small number of boilers. Without a larger body of data on which to base the emission factors, it is not known how the use of these factors biases the results.
- The exclusion of the inputs and outputs from materials extraction, the initial and secondary materials processing stages, and the associated transportation for several fuel types (see Table 2) will underestimate the total inventory from electricity generation. Especially when considering the massive amount of processing that is required to develop the uranium fuel pellets that are used in nuclear generation, this may be the largest influence of all the biases for this inventory. (It should also be noted that these processing steps for developing uranium pellets are extremely energy intensive, thus decreasing the efficiency of not only the nuclear electricity generation process, but also the whole grid's overall efficiency as well.)
- The exclusion of the hydropower and renewable generation types should have little effect on the U.S.-wide electric grid inventory data because the renewables are a minimal percentage of the total generation, and hydropower involves no combustion and thus would have minimal impacts.
- The exclusion of the nonutility-type generation brings two major issues forward. First, nonutilities use cogeneration the majority of the time to produce electricity (EIA 1997). Cogeneration utilizes two cycles to produce electrical energy and another form of usable energy (typically steam) thereby increasing the overall efficiency of the energy conversion process. Second, renewables (including hydropower) and gas make up 76% of the energy generation type for nonutilities, two of the lesser polluting types of electricity generation. These two factors should combine to, for the average kWh, decrease not only the quantity of raw materials needed as inputs but also the amount of air emissions generated as outputs for the inventory.

Finally, the Japanese inventory is limited by having been derived from the U.S. fuel-specific inventories and may not accurately represent electricity production operations in Japan.

Overall, with some implications inferring an underestimate in the inventory and some inferring an overestimate, it is uncertain which way the inventory may be biased. To a certain extent, the lessened impacts from the exclusion of nonutility electricity production data seem to offset the exclusion of the processing and transportation life-cycles substages data. However, the

exclusion of this relevant data hampers the effort to build a completely accurate and representative electric grid inventory.

7. CONCLUSIONS

The U.S. electric grid inventory data presented in this TM will be used to determine the environmental burdens that result from energy consumption in the U.S. using the CDP LCIA methodology. To summarize the work done in this analysis and its use within the CDP, the following is presented:

- The fuel-specific inventories for each of the four primary generation categories used in the average electric grid were compiled from a variety of sources, and then the fuel-specific inventories were combined using a generation-based weighted average to develop the U.S.-wide and Japanese electrical energy grid inventories of material inputs and pollutant outputs in units of grams of input or output per kWh consumed (except for radioactive materials which were placed in units of Becquerels/kWh).
- The U.S.-wide electrical energy grid inventory data are used in the CDP to calculate material inputs and pollutant outputs from electricity consumed during some manufacturing, use, and final disposal of computer displays.
- The Japanese electrical energy grid inventory data are used in the CDP to calculate material inputs and pollutant outputs from electricity consumed during the manufacturing of monitors in Asia.
- The electrical energy grid data are not used for upstream life-cycle stages (i.e., extraction of materials, materials processing) as these are being obtained from secondary sources which already have included the inputs and outputs from energy consumption.

Once the data have been gathered for all of the CDP processes of interest and the inputs and outputs for each of these processes have been analyzed from a life-cycle perspective, the results will help identify how important the impacts of energy consumption are throughout a monitor's life-cycle.

ACRONYMS/ABBREVIATIONS

CCPCT	Center for Clean Products and Clean Technologies
CEC	California Energy Commission
CH_4	Methane
CO	Carbon monoxide
CO_2	Carbon dioxide
DOE	Department of Energy
EIA	Energy Information Administration
EPA	Environmental Protection Agency
FGD	Flue Gas Desulfurization
GWh	Gigawatthour
HCl	Hydrogen chloride
HF	Hydrogen fluoride
kWh	kilowatthour
LCA	Life-cycle assessment
LCI	Life-cycle inventory
LLRW	Low-level radioactive waste
MW	Megawatt
Mwh	Megawatthour
N ₂ O	Nitrous oxide
NO _x	Oxides of nitrogen
ORNL	Oak Ridge National Laboratory
РАН	Polycyclic aromatic hydrocarbons
Pb	Lead
PM-10	Particulate matter 10 microns or less in diameter
PWR	Pressurized Water Reactor
SO_2	Sulfur dioxide
TNMOC	Total nonmethane organic carbon
TOC	Total organic carbon
TRI	Toxic Release Inventory
U_3O_8	Uranium oxide
UT	University of Tennessee

REFERENCES

- CEC (California Energy Commission). 1979. Technical Assessment Manual, Vol. 1, Electricity Generation. p. A-4.
- CFR (Code of Federal Regulations). 1995. Title 40: Protection of Environment, Part 261, Section 2 - Definition of Solid Waste. June.
- EIA (Energy Information Administration). 1996. Nuclear Power Generation and Fuel Cycle Report 1996, Volume II, DOE/EIA-0436(96). October.
- EIA. 1997. Electric Power Annual 1996, Volume II, DOE/EIA-0348(96)/2. November.
- EIA. 1997. *Country Energy Data Report for 1997*. Web site available at http://www.eia.doe.gov.
- EIA. 1998a. Electric Power Annual 1997, Volume II, DOE/EIA-0348(97)/2. October.
- EIA. 1998b. Uranium Industry Annual 1997, DOE/EIA-0478(97). April.
- EIA. 1998c. Electric Power Annual 1997, Volume I, DOE/EIA-0348(97)/1. June.
- EIA. 1999a. Electric Power Monthly, February 1999 (With Data for November 1998), DOE/EIA-0226(99/02).
- EIA. 1999b. *Short Term Energy Outlook, September 1999 (an EIA document published online)*, Web site available at http://www.eia.doe.gov/pubs/forecasting/steo/sep99/a8tab.pdf. September.
- EPA (Environmental Protection Agency). 1996. Compilation of Air Pollutant Emission Factors (AP-42), Fifth Edition, Volume I. AP-42. January.
- EPA. 1998a. Inventory of Greenhouse Gas Emissions and Sinks, 1990 1996, EPA/236-R-98-006, March. {Annex Q}
- EPA. 1998b. National Air Quality & Emissions Trends Report, 1997, EPA/454/R-98-016. December.
- EPA. 1998c. From the Excel spreadsheet "Coal_Inv_Sheet" of the Coalbed Methane Outreach Program, as provided by the Program's primary contact Karl Schultz.
- Loiselle, V. 1998. Personal communication with V. Loiselle, Past Chairman of the Association of Radioactive Metal Recyclers, and J. Overly, University of Tennessee, Center for Clean Products and Clean Technologies. May.

- NEI (Nuclear Energy Institute). 1997. Nuclear Energy's Clean Air Benefits: A Legacy of Greenhouse Gas Reductions and Fossil Fuel Conservation in America. Washington, D.C. October.
- ORNL (Oak Ridge National Laboratory). 1994. Report 3: Estimating Externalities of Coal Fuel Cycles. Oak Ridge, TN. September.
- ORNL. 1995. Report 8: Estimating Externalities of Nuclear Fuel Cycles. Oak Ridge, TN. April.

ATTACHMENT A. COAL INFORMATION WORKSPACE

Pollutant	Quantity [a]	Units	Converted Quantity [b]	Units
Outputs				
Methane	169,906	million ft 3/yr	1.52E+05	million ft 3/yr

Table A1. Mining Related Outputs - Methane Air Emissions

[a] Value pertains to methane emissions related to all coal mined nation-wide in U.S. (EPA 1998a).

[b] This value represents methane emissions from coal mined exclusively for electricity generation. It is calculated based on the knowledge that 89.6% of the coal mined in the U.S. in 1997 was for electricity generation, thus the toal methane emissions are multiplied by 89.6% to obtain the emissions from electricity generation alone. (EIA web page: http://www.eia.doe.gov/cneaf/coal/cia/summary/cia_sum.html, 1997).

Table A2. Generation Related Inputs and Outputs - Excluding Air Emissions

Material/Pollutant	Quantity	Units	Converted Quantity	Units					
Inputs									
Ancillary Materials									
Water [a]	0.49	gallons/kWh							
Limestone [b]	12,091,817	tons/yr							
Lime [b]	5,310,548	tons/yr							
Outputs									
SOLID WASTES [c]									
Dust/sludge	55.143	tons/GWh	1.10E+02	lbs/MWh					
Coal waste	142.857	tons/GWh	2.86E+02	lbs/MWh					
Fly/bottom ash	35.714	tons/GWh	7.14E+01	lbs/MWh					
WASTEWATER EM	WASTEWATER EMISSIONS [c]								
Dissolver	0.278	tons/GWh	5.56E-01	lbs/MWh					
Suspended solids	0.005	tons/GWh	1.00E-02	lbs/MWh					
Sulfate	0.192	tons/GWh	3.84E-01	lbs/MWh					

[a] CEC (1979).

[b] Primary data used to calculate these values from EIA (1997); however, data were modified to derive nationwide average.

[c] ORNL (1994).

Pollutant	Quantity (lbs/ton of coal)	EPA's Factor Raging [b]
Outputs	•	•
Miscellaneous Compounds [c]		
Methane	0.04	В
Nitrous oxide	0.03	В
Hydrogen chloride	1.2	В
Hydrogen fluoride	0.15	В
Total organic compounds	0.3	Е
Total nonmethane organic compounds	0.06	В
Speciated Organic Compounds [d], [e]		
Acetaldehyde	5.70E-04	С
Acetophenone	1.50E-05	D
Acrolein	2.90E-04	D
Benzene	1.30E-03	А
Benzyl chloride	7.00E-04	D
Bis(2-ethyhexyl)pththalate (DEHP)	7.30E-05	D
Bromoform	3.90E-05	Е
Carbon disulfide	1.30E-04	D
2-Chloroacetophenone	7.00E-06	Е
Chlorobenzene	2.20E-05	D
Chloroform	5.90E-05	Е
Cumene	5.30E-06	Е
Cyanide	2.50E-03	D
2,4-Dinitrotoluene	2.80E-07	D
Dimethyl sulfate	4.80E-05	Е
Ethyl benzene	9.40E-05	D
Ethyl chloride	4.20E-05	D
Ethylene dichloride	4.00E-05	Е
Ethylene dibromide	1.20E-06	Е
Formaldehyde	2.40E-04	А
Hexane	6.70E-05	D
Isophorone	5.80E-04	D
Methyl bromide	1.60E-04	D
Methyl chloride	5.30E-04	D
Methyl ethyl ketone	3.90E-04	D
Methyl hydrazine	1.70E-04	Е
Methyl methacrylate	2.00E-05	E
Methyl tert butyl ether	3.50E-05	Е
Methylene chloride	2.90E-04	D

Table A3. Generation Related Outputs - Air Emissions [a]

Pollutant	Quantity (lbs/ton of coal)	EPA's Factor Raging [b]
Phenol	1.60E-05	D
Propionaldehyde	3.80E-04	D
Tetrachloroethylene	4.30E-05	D
Toluene	2.40E-04	А
1,1,1-Trichloroethane	2.00E-05	Е
Styrene	2.50E-05	D
Xylenes	3.70E-05	С
Vinyl acetate	7.60E-06	Е
Trace Metals [d], [e]		
Antimony	1.80E-05	А
Arsenic	4.10E-04	А
Beryllium	2.10E-05	А
Cadmium	5.10E-05	А
Chromium	2.60E-04	А
Chromium (VI)	7.90E-05	D
Cobalt	1.00E-04	А
Magnesium	1.10E-02	А
Manganese	4.90E-04	А
Mercury	8.30E-05	А
Nickel	2.80E-04	А
Selenium	1.30E-03	А
Polycyclic Aromatic Hydrocarbons [d], [e]		
Biphenyl	1.70E-06	D
Acenaphthene	5.10E-07	В
Acenaphthylene	2.50E-07	В
Anthracene	2.10E-07	В
Benmzo(a)anthracne	8.00E-08	В
Benzo(a)pyrene	3.80E-08	D
Benzo(b,j,k)fluoranthene	1.10E-07	D
Benzo(g,h,i)perylene	2.70E-08	D
Chrysene	1.00E-07	С
Fluoranthene	7.10E-07	В
Fluorene	9.10E-07	В
Indeno(1,2,3-cd)pyrene	6.10E-08	С
Naphthalene	1.30E-05	С
Phenanthrene	2.70E-06	В
Pyrene	3.30E-07	В
5-Methyl chrysens	2.20E-08	D
Dioxins & Furans [d], [f]		

 Table A3. Generation Related Outputs - Air Emissions [a]

Pollutant	Quantity	EPA's Factor Raging [b]	
	(IDS/ton of coal)		
2,3,7,8-TCDD	1.43E-11	E	
Total TCDD	9.28E-11	D	
Total PeCDD	4.47E-11	D	
Total HxCDD	2.87E-11	D	
Total HpCDD	8.34E-11	D	
Total OCDD	4.16E-10	D	
Total PCDDd	6.66E-10	D	
2,3,7,8-TCDF	5.10E-11	D	
Total TCDF	4.04E-10	D	
Total PeCDF	3.53E-10	D	
Total HxCDF	1.92E-10	D	
Total HpCDF	7.68E-11	D	
Total OCDF	6.63E-11	D	
Total PCDFd	1.09E-09	D	
TOTAL PCDD/PCDF	1.76E-09	D	

 Table A3. Generation Related Outputs - Air Emissions [a]

[a] All the air emissions presented here are from EPA's AP-42 factors (EPA 1996).

[b] "EPA's AP-42 emissions factor rating is an overall assessment of how good a factor is, based on both the quality of the test(s) or information that is the source of the factor and on how well the factor represents the emission source." (EPA 1996) EPA's factor ratings are as follows:

A = Excellent; B = Above Average; C = Average; D = Below Average; and E = Poor.

[c] Due to a lack of data on firing configurations for all U.S. boilers, the pulverized coal (PC), dry, wall-fired boiler firing configuration (the most common configuration in the U.S.) was chosen as the representative configuration. Additionally, the factors for methane, nitrous oxide,, total organic compounds and total nonmethane organic compounds are for uncontrolled emissions, while the values for hydrogen chloride and hydrogen fluoride are for controlled and uncontrolled emissions (measurements were taken from different facilities were some had control equipment and others did not).

[d] These are all controlled factors.

[e] Apply to bituminous, subbituminous and ignite coal types. Even though these factors apply only to these three coal types, the emission factors are applied to all coal types. This was done for two reasons: 1) because no factors were given for the remaining coal types; and 2) those remaining coal types do generate some of these emissions, and it would have been erroneous to consider then 'pollutant-free."

[f] Apply to bituminous and subbituminous coal types. Even though these factors apply only to these two coal types, the emission factors are applied to all coal types. This was done for two reasons: 1) because no factors were given for the remaining coal types; and 2) those remaining coal types do generate some of these emissions, and it would have been erroneous to consider them 'pollutant-free.'

ATTACHMENT B. GAS INFORMATION WORKSPACE

Table B1. Generation Related Inputs

Material	Quantity		Units	
Inputs				
Ancillary Materials				
Water [a]		0.25	gallons/kWh	

[a] CEC (1979). The value listed under 'Combined Cycle' in the reference was used for gas-fired generation since the majority of combined cycle units's top cycle is fired by gas.

Table B2. Generation Related Outputs

Pollutant	Quantity (lbs/million ft3 of gas)	EPA's factor rating [b]
Outputs		
AIR EMISSIONS [a]		
Miscellaneous Compounds [c]		
Nitrous oxide	2.2	С
Total organic compounds	1.7	С
Speciated Organic Compounds [d]		
Fluoranthene	3.10E-06	E
Formaldehyde	1.55E-01	С
2-Methylnaphthalene	9.02E-06	Е
Naphthalene	2.40E-04	E
Phenanthrene	1.00E-05	Е
Pyrene	5.10E-06	Ε
Toluene	2.20E-03	Е
Trace Metals [e]		
Arsenic	2.30E-04	Е
Barium	2.40E-03	Е
Chromium	1.10E-03	Е
Cobalt	1.20E-04	Е
Copper	2.51E-04	Е
Manganese	3.81E-04	E
Molybdenum	5.81E-04	E
Nickel	3.61E-03	Е
Vanadium	3.21E-03	Е

[a] All outputs for gas-fired generation are air emissions and are from the EPA's AP-42 emission factors (EPA 1996).

[b] "EPA's AP-42 emissions factor rating is an overall assessment of how good a factor is, based on both the quality of the test(s) or information that is the source of the factor and how well the factor represents the emission source (EPA 1996). EPA's factor ratings are as follows: A = Excellent; B = Above Average; C = Average; D = Below Average; and E = Poor.

[c] Factors are for uncontrolled combustion.

[d] Each of the seven factors provided are for controlled and uncontrolled combustion.

[e] Factors are for controlled combustion.

ATTACHMENT C. PETROLEUM INFORMATION WORKSPACE

Material	Quantity	Units	
Inputs			
ANCILLARY MATERIALS			
Water [a]	0.43	gallons/kWh	

Table C1. Generation Related Inputs

[a] CEC (1979).

Table C2. Generation Related Outputs

Pollutant	Quantity (lbs/thousand gallons of oil)	EPA's factor rating [b]	Quantity (lbs/thousand gallons of oil)	EPA's factor rating [b]	Quantity (lbs/thousand gallons of oil)	EPA's factor rating [b]
	Fuel Oil #6		Fuel Oil #2		Average	Factors
Outputs						
AIR EMISSIONS [a]						
Miscellaneous Compoun	ds [c]					
Methane	0.28	А	0.052	А	2.66E-01	А
Nitrous oxide	0.11	В	0.11	В	1.10E-01	В
Total organic compounds	1.04	А	0.252	А	9.93E-01	А
Total nonmethane organic compounds	0.76	А	0.2	А	7.26E-01	A
Speciated Organic Com	oounds [d]					
Acenaphthene	2.11E-05	С			2.11E-05	С
Acenaphthylene	2.53E-07	D			2.53E-07	D
Anthracene	1.22E-06	С			1.22E-06	С
Benzene	2.14E-04	С			2.14E-04	С
Benz(a)anthracene	4.01E-06	С			4.01E-06	С
Benzo(b,k)fluoranthene	1.48E-06	С			1.48E-06	С
Benzo(g,h,i)perylene	2.26E-06	С			2.26E-06	С
Chrysene	2.38E-06	С			2.38E-06	С
Dibenzo(a,h)anthracene	1.67E-06	D			1.67E-06	D
Ethylbenzene	6.36E-05	Е			6.36E-05	Е
Fluoranthene	4.84E-06	С			4.84E-06	С
Fluorene	4.47E-06	С			4.47E-06	С
Formaldehyde	3.30E-02	С			3.30E-02	С
Indo(1,2,3-cd)pyrene	2.14E-06	С			2.14E-06	С
Naphthalene	1.13E-03	С			1.13E-03	С
OCDD	3.10E-09	Е			3.10E-09	Е
Phenanthrene	1.05E-05	С			1.05E-05	С
Pyrene	4.25E-06	С			4.25E-06	С
Toluene	6.20E-03	D			6.20E-03	D

Pollutant	Quantity (lbs/thousand gallons of oil)	EPA's factor rating [b]	Quantity (lbs/thousand gallons of oil)	EPA's factor rating [b]	Quantity (lbs/thousand gallons of oil)	EPA's factor rating [b]
	Fuel Oil #6		Fuel Oil #2		Average	Factors
1,1,1-Trichloroethane	2.36E-04	Е			2.36E-04	E
o-xylene	1.09E-04	Е			1.09E-04	Е
Trace Metals [e]	_	_	_	-	_	_
Antimony	5.25E-03	Е			5.25E-03	Е
Arsenic	1.32E-03	С	5.88E-04	Е	1.28E-03	С
Barium	2.57E-03	D			2.57E-03	D
Beryllium	2.78E-05	С	3.50E-04	Е	4.71E-05	С
Cadmium	3.98E-04	С	1.54E-03	Е	4.67E-04	С
Chloride	3.47E-01	D			3.47E-01	D
Chromium	8.45E-04	С	8.05E-03	Е	1.28E-03	С
Chromium (VI)	2.48E-04	С			2.48E-04	С
Cobalt	6.02E-03	D			6.02E-03	D
Copper	1.76E-03	С			1.76E-03	С
Fluoride	3.73E-02	D			3.73E-02	D
Manganese	3.00E-03	С	1.96E-03	Е	2.94E-03	С
Mercury	1.13E-04	С	4.20E-04	Е	1.31E-04	С
Molybdenum	7.87E-04	D			7.87E-04	D
Nickel	8.45E-02	С	2.38E-02	Е	8.09E-02	С
Phosphorous	9.46E-03	D			9.46E-03	D
Selenium	6.83E-04	С			6.83E-04	С
Vanadium	3.18E-02	D			3.18E-02	D
Zinc	2.91E-02	D			2.91E-02	D

 Table C2. Generation Related Outputs

[a] All outputs for oil-fired generation are air emissions and are from the EPA's AP-42 emission factors (EPA 1996). [b] "EPA's AP-42 emissions factor rating is an overall assessment of how good a factor is, based on both the quality of the test(s) or information that is the source of the factor and on how well the factor represents the emission source" (EPA 1996). EPA's factor ratings are as follows: A = Excellent; B = Above Average; C = Average; D = Below Average; and E = Poor.

[c] Factors are for uncontrolled combustion of all pollutants, except for N2O. No information was provided as to the control of N2O emissions.

[d] No information was provided as to the control of these pollutants. No emission factors were presented for fuel oil #2, thus, the factors for #6 were considered the same for #2. Even though one could assume that fuel oil #2 should have smaller quantities of pollutants to contribute per equivalent volume than fuel oil #6 (due to #6 being a residual oil and # 2 being a distillate), some of the results from the trace metals table contradict that assumption, thus substantiating this action.

[e] No information was provided as to the control of these pollutants. Where no fuel oil #2 emission factor was given, the factor for fuel oil #6 was used for both fuel oils. Even though one could assume that fuel oil #2 should have smaller quantities of pollutants to contribute per equivalent volume than fuel oil #6, some of the results from this table (where factors were reported for both fuel oils) contradict that assumption, thus substantiating this action.

Table C3. Miscellaneous Calculation Information

	Quantity	Unit	Quantity	Unit
	Fuel Oil #6		Fuel (Dil #2
Type of petroleum used at utilities [a]	94	%	6	%
AP-42 heat values for fuel oils [f]	152,000	btu/gal	140,000	but/gal

[a] Source: EIA's "Cost & Quality of Fuels for Electric Utility Plants 1997."[b] EPA (1996).

ATTACHMENT D. NUCLEAR INFORMATION WORKSPACE

Table D1. Generation Related Inputs and Outputs - Excluding Radionuclide Emissions

Material/Pollutant	Quantity	Units	Converted Quantity	Units
Inputs				
PRIMARY MATERIALS				
Uranium oxide [a]	48,700,000	lbs/yr		
ANCILLARY MATERIALS				
Water [b]	0.62	gallons/kWh		
Outputs				
SOLID/HAZARDOUS WASTES				
Uranium (spent fuel generated) [c]	2,400	metric tons/yr	5.29E+06	lbs/yr
Low-level radioactive waste [d]	220,500	cubic feet/yr		

[a] Known as 'yellowcake,' this post-milling uranium product is sent to conversion facilities (EIA 1998b). [b] CEC (1979).

[c] EIA (1996); Table 18. [d] Loisell (1998).

Table D2. Generation Related Outputs - Radionuclide Emissions

Isotope	Quantity	Units	Converted Quantity	Units [d]		
Outputs [a], [b], [c]						
AIRBORNE RADIONUCLIDE EMISSIONS						
T-3	5.98E+01	Curies/yr	7.33E-06	Curies/MWh		
Ar-41	2.55E+01	Curies/yr	3.13E-06	Curies/MWh		
Cr-51	1.60E-03	Curies/yr	1.96E-10	Curies/MWh		
Mn-54	2.27E-05	Curies/yr	2.78E-12	Curies/MWh		
Co-57	4.30E-06	Curies/yr	5.27E-13	Curies/MWh		
Co-58	5.49E-05	Curies/yr	6.73E-12	Curies/MWh		
Co-60	4.13E-04	Curies/yr	5.06E-11	Curies/MWh		
Kr-85	4.23E+01	Curies/yr	5.18E-06	Curies/MWh		
Kr-85M	2.05E+00	Curies/yr	2.51E-07	Curies/MWh		
Kr-87	7.63E-01	Curies/yr	9.35E-08	Curies/MWh		
Rb-88	8.38E-03	Curies/yr	1.03E-09	Curies/MWh		
Kr-88	3.58E+00	Curies/yr	4.39E-07	Curies/MWh		
Br-89	2.95E-06	Curies/yr	3.62E-13	Curies/MWh		
Br-90	1.20E-06	Curies/yr	1.47E-13	Curies/MWh		
Nb-95	9.02E-07	Curies/yr	1.11E-13	Curies/MWh		
Zr-95	2.33E-06	Curies/yr	2.86E-13	Curies/MWh		
Тс-99М	1.21E-07	Curies/yr	1.48E-14	Curies/MWh		
Ag-110M	2.69E-08	Curies/yr	3.30E-15	Curies/MWh		
I-131	1.93E-03	Curies/yr	2.37E-10	Curies/MWh		

Isotope	Quantity	Units	Converted Quantity	Units [d]
Xe-131M	3.45E+00	Curies/yr	4.23E-07	Curies/MWh
I-132	3.92E-04	Curies/yr	4.80E-11	Curies/MWh
Xe-133	4.98E+02	Curies/yr	6.10E-05	Curies/MWh
I-133	1.79E+00	Curies/yr	2.19E-07	Curies/MWh
Xe-133M	3.31E+01	Curies/yr	4.06E-06	Curies/MWh
Cs-134	8.10E-05	Curies/yr	9.93E-12	Curies/MWh
I-134	2.03E-03	Curies/yr	2.49E-10	Curies/MWh
Xe-135	1.88E+01	Curies/yr	2.30E-06	Curies/MWh
I-135	1.02E-04	Curies/yr	1.25E-11	Curies/MWh
Xe-135M	3.59E-01	Curies/yr	4.40E-08	Curies/MWh
Cs-137	6.11E-04	Curies/yr	7.49E-11	Curies/MWh
Xe-138	1.19E+00	Curies/yr	1.46E-07	Curies/MWh
T-3	4.47E+02	Curies/yr	5.48E-05	Curies/MWh
Na-24	2.24E-03	Curies/yr	2.75E-10	Curies/MWh
Cr-51	6.07E-02	Curies/yr	7.44E-09	Curies/MWh
Mn-54	4.00E-02	Curies/yr	4.90E-09	Curies/MWh
Fe-55	1.43E-01	Curies/yr	1.75E-08	Curies/MWh
Co-57	1.47E-03	Curies/yr	1.80E-10	Curies/MWh
Co-58	5.98E-01	Curies/yr	7.33E-08	Curies/MWh
Fe-59	7.34E-03	Curies/yr	9.00E-10	Curies/MWh
Co-80	1.57E-01	Curies/yr	1.92E-08	Curies/MWh
Zn-85	6.75E-04	Curies/yr	8.27E-11	Curies/MWh
Kr-85M	3.78E-02	Curies/yr	4.63E-09	Curies/MWh
Sr-89	2.42E-03	Curies/yr	2.97E-10	Curies/MWh
Sr-90	5.69E-04	Curies/yr	6.97E-11	Curies/MWh
Nb-95	1.03E-02	Curies/yr	1.26E-09	Curies/MWh
Sr-95	6.27E-03	Curies/yr	7.68E-10	Curies/MWh
Mo-99	7.55E+04	Curies/yr	9.25E-03	Curies/MWh
Tc-99M	8.78E-04	Curies/yr	1.08E-10	Curies/MWh
Ru-103	1.26E-03	Curies/yr	1.54E-10	Curies/MWh
Ag-110M	1.47E-02	Curies/yr	1.80E-09	Curies/MWh
Sn-113	1.39E-03	Curies/yr	1.70E-10	Curies/MWh
Sb-124	1.26E-02	Curies/yr	1.54E-09	Curies/MWh
Sb-125	5.02E-02	Curies/yr	6.15E-09	Curies/MWh
I-131	2.80E-02	Curies/yr	3.43E-09	Curies/MWh
Xe-131M	4.60E-01	Curies/yr	5.64E-08	Curies/MWh
I-132	1.06E-02	Curies/yr	1.30E-09	Curies/MWh
Xe-133	7.07E+01	Curies/yr	8.66E-06	Curies/MWh
I-133	1.20E-02	Curies/yr	1.47E-09	Curies/MWh

 Table D2. Generation Related Outputs - Radionuclide Emissions

Isotope	Quantity	Units	Converted Quantity	Units [d]
Xe-133M	5.79E-01	Curies/yr	7.10E-08	Curies/MWh
Cs-134	3.37E-02	Curies/yr	4.13E-09	Curies/MWh
I-135	8.61E-03	Curies/yr	1.06E-09	Curies/MWh
Xe-135	5.27E-01	Curies/yr	6.46E-08	Curies/MWh
s-136	1.35E-03	Curies/yr	1.65E-10	Curies/MWh
Cs-137	5.06E-02	Curies/yr	6.20E-09	Curies/MWh
Ba-140	9.34E-04	Curies/yr	1.14E-10	Curies/MWh
La-140	1.00E-03	Curies/yr	1.23E-10	Curies/MWh

 Table D2. Generation Related Outputs - Radionuclide Emissions

[a] ORNL (1995).

[b] There were 111 operating commercial nuclear power stations in the U.S. in 1995, of which 75 were PWRs (presssurized-water reactors) and 36 were BWRs (boiling-water reactors). Of the PWRs, 52 were designed by Westinghouse; the remaining 23 were designed by either ABB or Babcock & Wilcox. From the available data, the authors chose to use data from all Westinghouse PWRs with greater than 800 MWS capacity (36 facilities) in attempts to gather consistent data for

their calculations on transport and dose-effects for their reference PWR.

[c] "These emissions are characterisite of normal (nonaccident) emissions, and did not appear to readily correlate to MWhs produced or reactor capacity. Amounts produced are more likely affected by random events within reactor, such as fuel pin cladding failures, leaks in the primary coolant loop, steam-generator tube leaks, etc." (ORNL 1995, pp. 6-35 & 6-36).

[d] The units of Curies/MWh were derived by dividing the Curies/yr values by the number of MWhs produced annually in the virtual facility discussed in the source for this information (8,160,000 MWh/yr).

ATTACHMENT E. SUMMARY INFORMATION WORKSPACE

Notes:

• In the tables in this Appendix, "US kWh" is utilized to identify values that are representative of an average U.S. kWh in 1997.

• Unless stated otherwise, the sources for the data presented here are detailed in Attachments A through D.

Material/Pollutant	Quantity	Units	Converted Quantity	Units	
Inputs					
PRIMARY MATERIALS					
Coal	900,361,000	tons/yr	2.83E+02	grams/US kWh	
Gas	2,968,453,000,000	ft3/yr	2.20E+01	grams/US kWh	
Petroleum	5,256,132,000	agal/yr	5.99E+00	grams/US kWh	
Uranium (yellow cake)	48,700,000	lbs/yr	7.64E-03	grams/US kWh	
ANCILLARY MATERIA	LS				
Limestone	12,091,817	tons/yr	3.79E+00	grams/US kWh	
Lime	5,310,548	tons/yr	1.67E+00	grams/US kWh	
Water	11,426,836,328,400	lbs/yr	1.79E+03	grams/US kWh	
Outputs					
SOLID WASTES					
Dust/sludge	197,169,972,516	lbs/yr	3.09E+01	grams/US kWh	
Coal waste	510,801,203,484	lbs/yr	8.01E+01	grams/US kWh	
Fly/bottom ash	127,699,406,968	lbs/yr	2.00E+01	grams/US kWh	
Uranium (spent fuel generated)	1.83E-06	lbs/kWh	8.30E-04	grams/US kWh	
LLRW	6.10E-06	lbs/kWh	2.77E-03	grams/US kWh	
WATER RELEASES					
Dissolver	994,020,136	lbs/yr	1.56E-01	grams/US kWh	
Suspended solids	17,878,060	lbs/yr	2.80E-03	grams/US kWh	
Sulfate	686,517,504	lbs/yr	1.08E-01	grams/US kWh	

Table E1. Generation Kelated Inputs and Outputs - Excluding Air Emiss

Pollutant	Quantity	Units	Converted Quantity	Units		
Outputs						
Criteria Pollutants [a]						
Carbon dioxide	2,231,433,058	tons/yr	6.48E+02	grams/US kWh		
Nitrogen oxides	6,178,000	tons/yr	1.79E+00	grams/US kWh		
Sulfur dioxide	13,082,000	tons/yr	3.80E+00	grams/US kWh		
Carbon monoxide	406,000	tons/yr	1.18E-01	grams/US kWh		
Lead	64	tons/yr	1.86E-05	grams/US kWh		
Particulate matter (10 microns or less)	290,000	tons/yr	8.43E-02	grams/US kWh		
Uncategorized Pollutants						
Methane (includes coal mining releases)	6,473,160,931	lbs/yr	1.02E+00	grams/US kWh		
Nitrous oxide	34,119,601	lbs/yr	5.35E-03	grams/US kWh		
Hydrogen chloride	1,080,433,200	lbs/yr	1.70E-01	grams/US kWh		
Hydrogen fluoride	135,054,150	lbs/yr	2.12E-02	grams/US kWh		
Total organic compounds	280,372,537	lbs/yr	4.40E-02	grams/US kWh		
Total nonmethane organic compounds	57,839,714	lbs/yr	9.07E-03	grams/US kWh		
Speciated Organic Compounds						
Acetaldehyde	513,206	lbs/yr	8.05E-05	grams/US kWh		
Acetophenone	13,505	lbs/yr	2.12E-06	grams/US kWh		
Acrolein	261,105	lbs/yr	4.10E-05	grams/US kWh		
Benzene	1,171,594	lbs/yr	1.84E-04	grams/US kWh		
Benzo(b,k)fluoranthene	8	lbs/yr	1.22E-09	grams/US kWh		
Benzyl chloride	630,253	lbs/yr	9.89E-05	grams/US kWh		
Bis(2-ethylhexyl)phthalate (DEHP)	65,726	lbs/yr	1.03E-05	grams/US kWh		
Bromoform	35114.079	lbs/yr	5.51E-06	grams/US kWh		
2-Chloroacetophenone	6302.527	lbs/yr	9.89E-07	grams/US kWh		
Carbon disulfide	117046.93	lbs/yr	1.84E-05	grams/US kWh		
Chlorobenzene	19807.942	lbs/yr	3.11E-06	grams/US kWh		
Chloroform	53121.299	lbs/yr	8.33E-06	grams/US kWh		
Cumene	4771.9133	lbs/yr	7.49E-07	grams/US kWh		
Cyanide	2250902.5	lbs/yr	3.53E-04	grams/US kWh		
2,4-Dinitrotoluene	252.10108	lbs/yr	3.96E-08	grams/US kWh		
Dibenzo(a,h)anthracene	8.77774044	lbs/yr	1.38E-09	grams/US kWh		
Dimethyl sulfate	43217.328	lbs/yr	6.78E-06	grams/US kWh		
Ethyl benzene	84968.224	lbs/yr	1.33E-05	grams/US kWh		
Ethyl chloride	37815.162	lbs/yr	5.93E-06	grams/US kWh		
Ethylene dichloride	36014.44	lbs/yr	5.65E-06	grams/US kWh		
Ethylene dibromide	1080.4332	lbs/yr	1.70E-07	grams/US kWh		

 Table E2. Generation Related Outputs - Air Emissions

Pollutant	Quantity	Units	Converted Quantity	Units
Formaldehyde	849649.211	lbs/yr	1.33E-04	grams/US kWh
Hexane	60324.187	lbs/yr	9.46E-06	grams/US kWh
Indo(1,2,3-cd)pyrene	11.24812248	lbs/yr	1.76E-09	grams/US kWh
Isophorone	522209.38	lbs/yr	8.19E-05	grams/US kWh
2-Methylnaphthalene	26.77544606	lbs/yr	4.20E-09	grams/US kWh
Methyl bromide	144057.76	lbs/yr	2.26E-05	grams/US kWh
Methyl chloride	477191.33	lbs/yr	7.49E-05	grams/US kWh
Methyl ethyl ketone	351140.79	lbs/yr	5.51E-05	grams/US kWh
Methyl hydrazine	153061.37	lbs/yr	2.40E-05	grams/US kWh
Methyl methacrylate	18007.22	lbs/yr	2.83E-06	grams/US kWh
Methyl tert butyl ether	31512.635	lbs/yr	4.94E-06	grams/US kWh
Methylene chloride	261104.69	lbs/yr	4.10E-05	grams/US kWh
OCDD	0.016294009	lbs/yr	2.56E-12	grams/US kWh
Phenol	14405.776	lbs/yr	2.26E-06	grams/US kWh
Propionaldehyde	342137.18	lbs/yr	5.37E-05	grams/US kWh
Tetrachloroethylene	38715.523	lbs/yr	6.07E-06	grams/US kWh
Toluene	255205.255	lbs/yr	4.00E-05	grams/US kWh
1,1,1-Trichloroethane	19247.66715	lbs/yr	3.02E-06	grams/US kWh
Styrene	22509.025	lbs/yr	3.53E-06	grams/US kWh
Xylenes	33313.357	lbs/yr	5.23E-06	grams/US kWh
o-xylene	572.918388	lbs/yr	8.99E-08	grams/US kWh
Vinyl acetate	6842.7436	lbs/yr	1.07E-06	grams/US kWh
Trace Metals				
Antimony	43,801	lbs/yr	6.87E-06	grams/US kWh
Arsenic	376,538	lbs/yr	5.91E-05	grams/US kWh
Barium	20,633	lbs/yr	3.24E-06	grams/US kWh
Beryllium	19,155	lbs/yr	3.01E-06	grams/US kWh
Cadmium	48,371	lbs/yr	7.59E-06	grams/US kWh
Chloride	1,823,878	lbs/yr	2.86E-04	grams/US kWh
Chromium	244,073	lbs/yr	3.83E-05	grams/US kWh
Chromium (VI)	72,432	lbs/yr	1.14E-05	grams/US kWh
Cobalt	122,034	lbs/yr	1.91E-05	grams/US kWh
Copper	9,996	lbs/yr	1.57E-06	grams/US kWh
Fluoride	196,054	lbs/yr	3.08E-05	grams/US kWh
Magnesium	9,903,971	lbs/yr	1.55E-03	grams/US kWh
Manganese	457,748	lbs/yr	7.18E-05	grams/US kWh
Mercury	75,421	lbs/yr	1.18E-05	grams/US kWh
Molybdenum	5,861	lbs/yr	9.20E-07	grams/US kWh
Nickel	687,818	lbs/yr	1.08E-04	grams/US kWh
Phosphorous	49,723	lbs/yr	7.80E-06	grams/US kWh

 Table E2. Generation Related Outputs - Air Emissions

Pollutant	Quantity	Units	Converted Quantity	Units		
Selenium	1,174,059	lbs/yr	1.84E-04	grams/US kWh		
Vanadium	176,674	lbs/yr	2.77E-05	grams/US kWh		
Zinc	152,953	lbs/yr	2.40E-05	grams/US kWh		
Polynuclear Aromatic Hydrocarbons (PAHs)						
Biphenyl	1,531	lbs/yr	2.40E-07	grams/US kWh		
Acenaphthene	459	lbs/yr	8.94E-08	grams/US kWh		
Acenaphthylene	225	lbs/yr	3.55E-08	grams/US kWh		
Anthracene	189	lbs/yr	3.07E-08	grams/US kWh		
Benzo(a)anthracene	72	lbs/yr	1.46E-08	grams/US kWh		
Benzo(a)pyrene	34	lbs/yr	5.37E-09	grams/US kWh		
Benzo(b,j,k)fluoranthene	99	lbs/yr	1.55E-08	grams/US kWh		
Benzo(g,h,i)perylene	24	lbs/yr	5.68E-09	grams/US kWh		
Chrysene	90	lbs/yr	1.61E-08	grams/US kWh		
Fluoranthene	639	lbs/yr	1.06E-07	grams/US kWh		
Fluorene	819	lbs/yr	1.32E-07	grams/US kWh		
Indeno(1,2,3-cd)pyrene	55	lbs/yr	8.62E-09	grams/US kWh		
Naphthalene	11,705	lbs/yr	2.88E-06	grams/US kWh		
Phenanthrene	2,431	lbs/yr	3.95E-07	grams/US kWh		
Pyrene	297	lbs/yr	5.25E-08	grams/US kWh		
5-Methyl chrysene	20	lbs/yr	3.11E-09	grams/US kWh		
Dioxins & Furans						
2,3,7,8-TCDD	0.012875162	lbs/yr	2.02E-12	grams/US kWh		
Total TCDD	0.083553501	lbs/yr	1.31E-11	grams/US kWh		
Total PeCDD	0.040246137	lbs/yr	6.31E-12	grams/US kWh		
Total HxCDD	0.025840361	lbs/yr	4.05E-12	grams/US kWh		
Total HpCDD	0.075090107	lbs/yr	1.18E-11	grams/US kWh		
Total OCDD	0.374550176	lbs/yr	6.13E-11	grams/US kWh		
Total PCDDd	0.599640426	lbs/yr	9.41E-11	grams/US kWh		
2,3,7,8-TCDF	0.045918411	lbs/yr	7.20E-12	grams/US kWh		
Total TCDF	0.363745844	lbs/yr	5.71E-11	grams/US kWh		
Total PeCDF	0.317827433	lbs/yr	4.99E-11	grams/US kWh		
Total HxCDF	0.172869312	lbs/yr	2.71E-11	grams/US kWh		
Total HpCDF	0.069147725	lbs/yr	1.08E-11	grams/US kWh		
Total OCDF	0.059693934	lbs/yr	9.37E-12	grams/US kWh		
Total PCDFd	0.98139349	lbs/yr	1.54E-10	grams/US kWh		
TOTAL PCDD/PCDF	1.58463536	lbs/yr	2.49E-10	grams/US kWh		

 Table E2.
 Generation Related Outputs - Air Emissions

[a] The criteria pollutants were not multiplied by the transmission and distribution factor due to their being reported totals for the electric industry.

Isotope	Quantity	Units	Converted Quality	Units
Outputs		-	-	
Airborne Radionuclides				
T-3	4606.974412	Curies/yr	5.90E+01	Becquerels/US kWh
Ar-41	1964.5125	Curies/yr	2.51E+01	Becquerels/US kWh
Cr-51	0.123263529	Curies/yr	1.58E-03	Becquerels/US kWh
Mn-54	0.001748801	Curies/yr	2.24E-05	Becquerels/US kWh
Co-57	0.000331271	Curies/yr	4.24E-06	Becquerels/US kWh
Co-58	0.00422948	Curies/yr	5.41E-05	Becquerels/US kWh
Co-60	0.031817399	Curies/yr	4.07E-04	Becquerels/US kWh
Kr-85	3258.779559	Curies/yr	4.17E+01	Becquerels/US kWh
Kr-85M	157.9313971	Curies/yr	2.02E+00	Becquerels/US kWh
Kr-87	58.78129559	Curies/yr	7.52E-01	Becquerels/US kWh
Rb-88	0.645592735	Curies/yr	8.26E-03	Becquerels/US kWh
Kr-88	275.8021471	Curies/yr	3.53E+00	Becquerels/US kWh
Br-89	0.000227267	Curies/yr	2.91E-06	Becquerels/US kWh
Br-90	9.24476E-05	Curies/yr	1.18E-06	Becquerels/US kWh
Nb-95	6.94898E-05	Curies/yr	8.89E-07	Becquerels/US kWh
Zr-95	0.000179503	Curies/yr	2.30E-06	Becquerels/US kWh
Tc-99M	9.3218E-06	Curies/yr	1.19E-07	Becquerels/US kWh
Ag-110M	2.07237E-06	Curies/yr	2.65E-08	Becquerels/US kWh
I-131	0.148686632	Curies/yr	1.90E-03	Becquerels/US kWh
Xe-131M	265.7869853	Curies/yr	3.40E+00	Becquerels/US kWh
I-132	0.030199565	Curies/yr	3.86E-04	Becquerels/US kWh
Xe-133	38365.77353	Curies/yr	4.91E+02	Becquerels/US kWh
I-133	137.9010735	Curies/yr	1.76E+00	Becquerels/US kWh
Xe-133M	2550.014265	Curies/yr	3.26E+01	Becquerels/US kWh
Cs-134	0.006240216	Curies/yr	7.99E-05	Becquerels/US kWh
I-134	0.156390603	Curies/yr	2.00E-03	Becquerels/US kWh
Xe-135	1448.346471	Curies/yr	1.85E+01	Becquerels/US kWh
I-135	0.00785805	Curies/yr	1.01E-04	Becquerels/US kWh
Xe-135M	27.65725441	Curies/yr	3.54E-01	Becquerels/US kWh
Cs-137	0.04707126	Curies/yr	6.02E-04	Becquerels/US kWh
Xe-138	91.67725	Curies/yr	1.17E+00	Becquerels/US kWh
Waterborne Radionuclides				
T-3	34436.74853	Curies/yr	4.41E+02	Becquerels/US kWh
Na-24	0.172568941	Curies/yr	2.21E-03	Becquerels/US kWh
Cr-51	4.676310147	Curies/yr	5.98E-02	Becquerels/US kWh
Mn-54	3.081588235	Curies/yr	3.94E-02	Becquerels/US kWh
Fe-55	11.01667794	Curies/yr	1.41E-01	Becquerels/US kWh
Co-57	0.113248368	Curies/yr	1.45E-03	Becquerels/US kWh
Co-58	46.06974412	Curies/yr	5.90E-01	Becquerels/US kWh
Fe-59	0.565471441	Curies/yr	7.24E-03	Becquerels/US kWh

Table E3. Generation Related Outputs - Radionuclides
Isotope	Quantity	Units	Converted Quality	Units
Co-80	12.09523382	Curies/yr	1.55E-01	Becquerels/US kWh
Zn-85	0.052001801	Curies/yr	6.65E-04	Becquerels/US kWh
Kr-85M	2.912100882	Curies/yr	3.73E-02	Becquerels/US kWh
Sr-89	0.186436088	Curies/yr	2.39E-03	Becquerels/US kWh
Sr-90	0.043835593	Curies/yr	5.61E-04	Becquerels/US kWh
Nb-95	0.793508971	Curies/yr	1.02E-02	Becquerels/US kWh
Sr-95	0.483038956	Curies/yr	6.18E-03	Becquerels/US kWh
Mo-99	5816497.794	Curies/yr	7.44E+04	Becquerels/US kWh
Tc-99M	0.067640862	Curies/yr	8.66E-04	Becquerels/US kWh
Ru-103	0.097070029	Curies/yr	1.24E-03	Becquerels/US kWh
Ag-110M	1.132483676	Curies/yr	1.45E-02	Becquerels/US kWh
Sn-113	0.107085191	Curies/yr	1.37E-03	Becquerels/US kWh
Sb-124	0.970700294	Curies/yr	1.24E-02	Becquerels/US kWh
Sb-125	3.867393235	Curies/yr	4.95E-02	Becquerels/US kWh
I-131	2.157111765	Curies/yr	2.76E-02	Becquerels/US kWh
Xe-131M	35.43826471	Curies/yr	4.54E-01	Becquerels/US kWh
I-132	0.816620882	Curies/yr	1.05E-02	Becquerels/US kWh
Xe-133	5446.707206	Curies/yr	6.97E+01	Becquerels/US kWh
I-133	0.924476471	Curies/yr	1.18E-02	Becquerels/US kWh
Xe-133M	44.60598971	Curies/yr	5.71E-01	Becquerels/US kWh
Cs-134	2.596238088	Curies/yr	3.32E-02	Becquerels/US kWh
I-135	0.663311868	Curies/yr	8.49E-03	Becquerels/US kWh
Xe-135	40.599925	Curies/yr	5.20E-01	Becquerels/US kWh
s-136	0.104003603	Curies/yr	1.33E-03	Becquerels/US kWh
Cs-137	3.898209118	Curies/yr	4.99E-02	Becquerels/US kWh
Ba-140	0.071955085	Curies/yr	9.21E-04	Becquerels/US kWh
La-140	0.077039706	Curies/yr	9.86E-04	Becquerels/US kWh

Table E3. Generation Related Outputs - Radionuclides

Generation type [a]	Net Generation						
	(MWh/yr)	%					
Coal	1,787,806,000	57.255%					
Gas	283,625,000	9.083%					
Petroleum	77,753,000	2.490%					
Nuclear	628,644,000	20.133%					
Hydro [b]	337,233,000	10.800%					
Renewables [b]	7,462,000	0.239%					
Total	3,122,522,000	100.000%					

Table E4. Net Generation Information

[a] This breakdown excludes nonutility electricity generation (Non and Independent Power Producers (NPPs or IPPs)), which typically contribute about 11% of the U.S. total (EIA 1999).

[b] Hydro and renewables were excluded from the calculation of inputs and outputs.

Table E5. Conversion Factors

Universal factor	1 pound =	453.59	grams
Water	1 gallon =	8.34	pounds
Coal [a]	1 pound =	10,275	btu
Gas [b]	a cubic foot =	0.0473	pounds
Petroleum	1 gallon =	7.26	pounds
Nuclear	1 Curie =	3.70E+10	Becquerals
Nuclear (LLRW)	1 cubic foot =	80	pounds
Transmission & distribution [c]	1 kWh out requires =	1.08	kWhs in

[a] Source: EIA's "Cost & Quality of Fuels for Electric Utility Plants 1997," Table ES-4.

[b] Calculated by the CCPCT with reference information from "Perry's Chemical Engineer's Handbook, 6th Edition."

[c] Source: EIA's Short-Term Energy Outlook," Table A8.

APPENDIX F

MANUFACTURING DATA COLLECTION QUESTIONNAIRE

this page intentionally left blank



DESIGN FOR THE ENVIRONMENT COMPUTER DISPLAY PROJECT Life-Cycle Inventory (LCI) Data Collection Questionnaire



Introduction

The Design for the Environment (DfE) Program in the U.S. Environmental Protection Agency's (EPA) Office of Pollution Prevention and Toxics has begun a voluntary, cooperative project with the electronics industry to assess the life-cycle environmental impacts of cathode ray tube (CRT) and liquid crystal display (LCD) desktop monitors. The DfE Program conducts comparative analyses of alternative products or processes to provide businesses with data to make environmentally informed choices about product or process improvements. The DfE Program has no regulatory or enforcement agenda and was established to act as a partner with industry to promote pollution prevention. This environmental life-cycle assessment will address human and ecological risk, energy and natural resource use, performance, and cost of various display technologies. The University of Tennessee (UT) Center for Clean Products and Clean Technologies is conducting the life-cycle inventory (LCI), which is the data collection phase of a life-cycle assessment, with technical assistance from the Asian Technology Information Program. Microelectronics and Computer Technology Corporation, the Electronics Industry Alliance, and other partners.

Boundaries

A life-cycle assessment considers impacts from materials acquisition, material manufacturing, product manufacturing, use, and final disposition of a product. The LCI data are intended to be used to evaluate relative environmental impacts over the entire life-cycle of a product, including transport between life-cycle stages. In this project, the product is either a color CRT or LCD monitor. Therefore, data associated with the materials and processes used directly in the manufacturing, use, and disposition of the product are relevant to the LCI and requested in this questionnaire. You will not need to include materials or energy not directly used in the production of the monitor or its components (e.g., general building heating and air conditioning).

Product focus

This project will evaluate CRT and LCD (twisted nematic and in-plane switching) technologies, based on <u>1998</u> production for a 17" CRT and a 15" LCD desktop monitor, with the following approximate specifications:

1024×768 resolution 100:1 contrast ratio 200 cd/m² brightness 262,000 colors



Fig. 1. Manufacturing process inventory conceptual template

Inventory data

We are asking you for data on a particular "product of interest"

that you manufacture, which is defined as a material, component, or subassembly that is part of the product focus defined above. The inputs and outputs data (Fig. 1) that you provide will be aggregated in the LCI to quantify the overall inputs and outputs of a CRT and LCD. Additionally, transportation information is requested in the inventory.

Data sources

Much of the requested information can be drawn from existing sources, including, but not limited to the following:

- 1. Purchase and production records
- 2. Bills and invoices
- 3. Material Safety Data Sheets (MSDS)
- 4. Toxic Release Inventory (TRI) forms

Audit and analysis results (e.g., wastewater discharge analyses)
 Local, state, and federal reporting forms (e.g., hazardous waste manifests)
 Local, state, and federal permits
 Monthly utility billing records

How the data will be used

<u>UT will aggregate the inventory data and tally the average inputs and outputs for the different monitors</u>. Information gathered by this questionnaire will be used to develop environmental profiles based on inputs and outputs for each stage in the manufacture of displays. The profiles will be used to evaluate environmental impacts from each product. Cost data will also be collected and presented along with environmental results. The environmental profiles can be used to encourage product design changes for product improvement. UT will aggregate data and ensure that data associated with particular companies remain anonymous to the EPA. UT can enter into confidentiality agreements where proprietary data are concerned. Please understand that accurate and representative information from you is critical for the success of this project.

Results of project

The results are intended to provide industry with an analysis of the life-cycle environmental impacts, cost, and performance of CRT and LCD computer monitors. Results will help identify areas for product and process improvement as related to risk and environmental impact (e.g., identifying material use inefficiencies) and will identify impacts from various life-cycle stages of the product systems. Use of the results will also help meet growing global demands of extended product responsibility.

Benefits of involvement

Your input will allow for your interests to be considered in the project development and data collection. By supplying data, the results will partially reflect your operations and, therefore, the results will be directly relevant to your interests. The project will allow you to directly apply results to your manufacturing process and identify areas for improvement. You will also be recognized as working voluntarily and cooperatively with the U.S. EPA.

Deadline

We are attempting to obtain all completed questionnaires before October 29, 1999.

Your cooperation and assistance are greatly appreciated.

For any questions, please contact Maria Leet Socolof at 423-974-9526, <socolofml@utk.edu> or Jonathan G. Overly at 423-974-3625, <jgoverly@utk.edu> at the University of Tennessee, 311 Conference Center Bldg., Knoxville, TN 37996-4134. Fax: 423-974-1838. For more project details, see the Project Fact Sheet, DfE Website <www.epa.gov/dfe>, or the Draft Final Goal Definition and Scoping Document.

INSTRUCTIONS

1. Please be sure to read the introductory text on each page before filling out the questionnaire.

Mar

- 2. The data you supply in the tables should represent inputs and outputs associated only with the "product of interest" (i.e., materials, components or subassemblies you manufacture that are either part of, or that are itself, the desktop monitor as defined on p. i under Product focus). If quantities provided are not specific to the "product of interest," please explain how they differ in the comments section at the bottom of the appropriate table.
- 3. Where supporting information is available as independent documents, reports or calculations, please provide them as attachments with reference to the associated page(s) or table(s) in this questionnaire.

4. If you have more than one product of interest to this project, please duplicate this questionnaire and fill out one questionnaire for each product.

5. If there is not adequate room on a page to supply your data (including comments), please copy the appropriate page and attach it to this packet.

6. The ensuing pages refer to the four indices shown below to detail specific information about the data. Additional information is provided below as required. <u>Data Quality Indicators Index</u>: These indicators will be used to assess the level of data quality in this questionnaire. Please report a DQI for the numerical value requested in each table on the following pages. The first category, Measured, pertains to a value that is a directly measured quantity. The second category, Calculated, refers to a value that required one or more calculations to obtain. The third category, Estimated, refers to a value that required a knowledgable employee's professional judgement to estimate. Lastly, the fourth category, Assumed, should be used only when a number had to be guessed.

Hazardous and Nonhazardous Waste Management Methods Index: These methods are applicable to both hazardous and nonhazardous wastes (Tables 7a and 7b). Please give the appropriate abbreviation in the Management Method column on p. 7 where requested. Depending on whether the management method is on or offsite, please indicate by specifying "on" or "off" in the appropriate column on p. 7.

m 11 o o o) / g 17		Fight Tables 2 - 21 - 1 - 5 - 6 - 7 - and 7 h
or lables 2, 3a, 3b, 4, /a, and /b:		ror ladies 5a, 50, 4, 5, 6a, 7a, and 70.
ransportation Modes Index		Data Quality Indicators Index
A - Large truck (18-wheeler), diesel		M - Measured
B - Small truck, diesel		C - Calculated
C - Small truck, gasoline		E - Estimated
D - Rail, diesel		A - Assumed
E - Barge, diesel		
F - Ocean freighter, diesel		
G - Other (please specify in comments section)		For Tables 7a and 7b:
		Hazardous and Nonhazardous Waste Management Methods Index
or Table 6b:		RU - Reused
astewater Treatment/Disposal Methods Index		R - Recycled
A - Direct discharge to surface water		L - Landfilled
B - Discharge to offsite wastewater treatment facility		Iv - Incinerated - volume reduction
C - Underground injection		Ic - Incinerated - energy conversion
D - Surface impoundment (e.g., settling pond)		S - Solidified/stabilized
E - Direct discharge to land		D - Deep well injected
F - Other (please specify in comments section)		O - Other (please specify in comments section)
····· /P······ ···· ···· ··········		hanne and a second s
	TE OUESTIC	ANG DI FASE CONTACT FITHER.
IF YOU HAV	LE QUESTIC	JINO, FLEADE CONTACT EITHEN.
- L. Barala (/Daviant) (Junana), Dhamar 422,024,0526	OP	Investion G. Overly (Project Engineer): Phone: 423-974-3625
a L. Socoloi (Project Manager). Phone: 423-974-9520	OK	Fav. 423-974-1838
rax: 423-974-1838		Paralle incontraction
Email: socolofini@utk.edu		itmail: igoveriy(wurk.edu

OFFICE OF MANAGEMENT AND BUDGET STATEMENT

The public reporting and recordkeeping burden for this collection of information is estimated to average 8 hours per response. Burden means the total time, effort, or financial resources expended by persons to generate, maintain, retain, or disclose or provide information to or for a Federal agency. The burden for this collection includes the time needed to review instructions; search data sources; complete and review the collection of information; and transmit or otherwise disclose the information. An agency may not conduct or sponsor, and a person is not required to respond to a collection of information unless it displays a currently valid OMB control number. The OMB control number for this collection is 2070-0152.

Send comments on the Agency's need for this information, the accuracy of the provided burden estimates, and any suggested methods for minimizing respondent burden, including through the use of automated collection techniques to the Director, OPPE Regulatory Information Division, U.S. Environmental Protection Agency (2136), 401 M Street, S.W., Washington, DC 20460. Include the OMB control number in any correspondence.

fable 1.	Facility Information	Conta	ct Information
1. Company name:		5a. Prepared by:	Date:
2. Facility name:		5b. Title:	
3. Facility address (location):		5c. Phone number:	Ext.:
		5d. Fax number:	
		Se Email address:	
	<u> </u>		
<u></u>	······································		
A Droducts manufactured oneite:			
4. Flocuets manufactured onsite.			
•••••••••••••••••••••••••••••••••••••••	n ing ang mananan ali a manila na mang kang pang pang kang di kang pang banang pang pang pang pang pang pang pa		
	Gener	c Data Collection Form - p. 1	

1. FACILITY & CONTACT INFORMATION

. Product of interest:			6. Energy consumption info	rmation (fill out only if the product of inter	est is an assembled
. 1998 annual production			- Active use:		watts
(e.g., units, kg, lbs):	an a		- Standby:		watts
 Facility's percent global market share for product of interest; 			- Active off:		watts
Product of interest unit weight:			 7. Performance specification Maximum resolution: Colors (at max. reso.): 	ns (fill out only if the product of interest is	an assembled monitor
. 1998 product of interest retail unit price:			Contrast ratio:Brightness:		cd/m²
Brief description of the main operati required to manufacture the product	ons/subprocesses of interest:				
Brief description of the main operative required to manufacture the product Product transport information. In the See Transportation Modes Index on	ons/subprocesses of interest: s table, please list the (b. iii for modes abbrevi	lop five locations (by quantity) to w ations. Percent capacity represents	hich you send the product of what percent of the transport	interest. You may supply one-way distance t vehicle's total load was carrying the produ	s in lieu of location's
Brief description of the main operati required to manufacture the product Product transport information. In the Sce Transportation Modes Index on Location (City, State, Country)	ons/subprocesses of interest: s table, please list the b. iii for modes abbrevi Distance	top five locations (by quantity) to w ations. Percent capacity represents Percent of production	hich you send the product of what percent of the transport Mode	interest. You may supply one-way distance t vehicle's total load was carrying the produ Number of trips annually	s in lieu of locations cts of interest. Percent capac
Brief description of the main operati required to manufacture the product Product transport information. In the Sce Transportation Modes Index on Location (City, State, Country)	ons/subprocesses of interest: s table, please list the (p. iii for modes abbrevi Distance	top five locations (by quantity) to w ations. Percent capacity represents Percent of production	hich you send the product of what percent of the transport Mode	interest. You may supply one-way distance t vehicle's total load was carrying the produ Number of trips annually	is in lieu of location's cts of interest. Percent capac
Brief description of the main operati required to manufacture the product Product transport information. In the See Transportation Modes Index on Location (City, State, Country)	ons/subprocesses of interest: s table, please list the t b. iii for modes abbrevi Distance	lop five locations (by quantity) to w ations. Percent capacity represents Percent of production	hich you send the product of what percent of the transport Mode	interest. You may supply one-way distance t vehicle's total load was carrying the produ Number of trips annually	s in lieu of location's cts of interest. Percent capac
Brief description of the main operati required to manufacture the product Product transport information. In the See Transportation Modes Index on Location (City, State, Country)	ons/subprocesses of interest: s table, please list the (p. iii for modes abbrevi Distance	lop five locations (by quantity) to w ations. Percent capacity represents Percent of production	hich you send the product of what percent of the transport Mode	interest. You may supply one-way distance t vehicle's total load was carrying the produ Number of trips annually	s in lieu of locations. cts of interest. Percent capac
Brief description of the main operati required to manufacture the product Product transport information. In the See Transportation Modes Index on Location (City, State, Country)	ons/subprocesses of interest: s table, please list the t b. iii for modes abbrevi Distance	lop five locations (by quantity) to w ations. Percent capacity represents Percent of production	hich you send the product of what percent of the transport Mode	interest. You may supply one-way distance t vehicle's total load was carrying the produ Number of trips annually	is in lieu of location cts of interest. Percent capa

2. PRODUCT OF INTEREST INFORMATION

.

F-5

3. PRIMARY & ANCILLARY INPUTS

- 1. <u>Primary & Ancillary Materials</u>: Primary materials, are defined as those materials that become part of the final product. Ancillary materials are those material inputs that assist production, yet do not become part of the final product. Please include the trade name and the generic name of each material where applicable.
- 2. CAS # or MSDS: Please either supply the chemical CAS (Chemical Abstract Service) number or attach a material "MSDS" to this document.
- 3. Annual quantity/units & Density/units: Please specify the amount of material consumed in 1998. Please use the units of mass-per-year (e.g., kg/yr, lb/yr). If you specify units of volume in lieu of mass, please provide the density.
- 4. Data quality indicators: See the Data Quality Indicators Index on p. iii for abbreviations. Please supply the DQI for the annual quantity value given.
- 5. <u>Recycled content</u>: Please specify the recycled content of each material identified. For example, 60/40/0 would represent a material that has 60% virgin material, 40% pre-consumer recycled and 0% post-consumer recycled content. Enter N/A (not applicable) for all components that are assemblies.
- 6. <u>Transportation information</u>: See the Transportation Modes Index on p. iii for mode abbreviations. Please specify the one-way transportation distance, or the location from where the material is shipped, and the number of trips made to your facility on an annual basis. % capacity represents what percent of the transport vehicle's total *load* was carrying the materials of interest.

Ta	ble 3a.	CAS#	Annual	Units	Density ³	Units	DQI	Recycled	Transportation I	nformat	ion (Rece	iving) ⁶
	Primary Materials ¹	or MSDS ²	Quantity ³				4	Content ⁵	Dist. or Location	Mode	# trips	% cap.
EX	AMPLE: 'GRTX resin (polypropylene resin)	MSDS	450,000	kg/yr			М	60/40/0	450 km	A	24	40
1.												
2.												
3.												
4.												
5.												
6.									1943 1			
7.							-			•		

Primary material comments:

2.

Table 3b.	CAS#	Annual	Units	Density ³	Units	DQI	Recycled	Transportation I	nformat	ion (Rece	iving) ⁶
Ancillary Materials ¹	or MSDS ²	Quantity ³				4	Content ⁵	Dist. or Location	Mode	# trips	% cap
EXAMPLE: Petroleum naphtha (cleaning solvent)	8032-32-4	920	liters/yr	0.96	kg/liter	C	100/0/0	St. Louis, MO	C	2	100
1.											
2.											
3.											
4.											
5.											
6.											
7.										- <u>.</u>	
Ancillary material comments:		· · ·									
				•							

4. UTILITY INPUTS

1. <u>Annual quantity/units</u>: Please specify the amount of each utility consumed in 1998. If possible, please exclude nonprocess-related consumption. If not possible, please include a comment that nonprocess-related consumption is included.

2. Data quality indicators: See the Data Quality Indicators Index on p. iii for abbreviations. Please supply the DQI for the annual quantity value given.

3. Transportation information: See the Transportation Modes Index on p. iii for mode abbreviations. Please specify the one-way transportation distance, or the location from where the fuel

is shipped, and the number of trips made to your facility on an annual basis. % capacity represents what percent of the transport vehicle's total *load* was carrying the materials of interest. 4. Individual Utility Notes:

Electricity:

The quantity of electricity should reflect only that used toward manufacturing the product of interest (identified on p. 2). One approach would be to start with your facility's total annual electrical energy consumption, estimate and remove nonprocess-related consumption, then estimate what portion of the remaining consumption is related to the specific operations of interest (if you manufacture more than one product). Please include consumption in all systems that use electricity for process-related purposes. Some examples include compressed air, chilled water, water deionization and HVAC consumption where clean or controlled environments are utilized.

Natural gas and LNG:

Please exclude all use for space heating or other nonprocess-related uses. If you choose to use units other than MCF (thousand cubic feet), please utilize only units of energy content or volume (e.g., mmBTU, therm, CCF).

Fuel oils:

Please use units of either volume or energy content (e.g., liters, cubic meters, mmBTU, MJ). Additionally, if the fuel oil is delivered by pipeline, enter "pipeline" in the Transportation Information space; if not delivered by pipeline, please include the associated transportation information.

All waters (e.g., deionized, city):

Please include all waters received onsite. Please indicate consumption in units of mass or volume.

Table 4	Annual U			Transportation Information (Receiving) ³				
Utilities ⁴	Quantity ¹			Dist. or Location	Mode	# trips	% сар.	
1. Electricity		MJ						
2. Natural gas		MCF						
3. Liquified natural gas (LNG)		MCF						
4. Fuel oil - type #2 (includes distillate and diesel)		liters						
5. Fuel oil - type #4		liters	_					
6. Fuel oil - type #6 (includes residual)		liters						
7. Propane		liters						
8. Water		liters						
9.								
10.						17		
11.								
12.								
13.								
Utility comments:								
		•						
					s 1			
		· · · ·						

5. AIR EMISSIONS

1. <u>Air emissions</u>: The emissions listed in the table below are some of the more common ones found in air release inventories; if you have information on other specific emissions, please include that information in the space provided. If you have any reporting forms or other air emission records for 1998, please attach copies to this questionnaire. Also, if you have information on stack as well as fugitive emissions, please copy this page and place each set of emissions on a different page. The energy consumed in any equipment used onsite to treat air emissions should be included in the utilities values on p. 4.

3. Data quality indicators: See the Data Quality Indicators Index on p. iii for abbreviations. Please supply the DQI for the annual quantity value given.

Table 5.	CAS	Annual	Units	DQI	Table 5 (continued).	CAS	Annual	Units	DQI
Air Emissions ¹	number	Quantity ²		3	Air Emissions ¹	number	Quantity		ļ
Total particulates					Ammonia	7664-41-7			-
Particulates < 10 microns (PM-10)					Arsenic	7440-38-2			
Sulfur oxides (SOx)				· · · · ·	Chromium	7440-47-3			
Nitrogen oxides (NOx)					Соррег	7440-50-8			
Carbon monoxide	630-08-0				Lead	7439-92-1			
Carbon dioxide	= 124-38-9				Manganese	7439-96-5			
Methane	74-82-8				Мегсигу	7439-98-7			
Benzene	71-43-2				Nickel	7440-02-0			
Toluene	108-88-3				Other emissions:				
Xylenes	1330-20-7				<u>l.</u>				
Naphthalene	91-20-3				2.				-
Total nonmethane VOCs					3				
Other speciated hydrocarbon emissions:					4				
1.					5.				<u> </u>
2.	,				6.				
3.					7				
4.					8.				
5.					9.				<u> </u>
6.					10.	<u>l</u>			
7.					Air emission comments:				
8.									
9.									
10.									
11.				<u> </u>					
12.									

^{2. &}lt;u>Annual quantity/units</u>: Please specify the amount of air emissions generated in 1998. If you do not have 1998 emissions data, use the next closest year's data you have and specify what year's data you are supplying in the comment section below. Please use units of mass-per-year (e.g., kg/yr, lb/yr).

6. WASTEWATER RELEASES & CONSTITUENTS

- 1. Annual quantity/units: Please specify the amount of wastewater(s) generated in 1998. Please use units of mass-per-year (e.g., kg/yr, lb/yr). If multiple streams exist, please copy this page and fill it out for each stream.
- 2. Data quality indicators: See the Data Quality Indicators Index on p. iii for abbreviations. Please include one DQI for the annual wastewater stream quantity value supplied, and one DQI for the wastewater constituents information supplied. If more than one DQI is applicable to the wastewater constituents data, please clarify this in the comment section.
- 3. <u>Wastewater constituents</u>: Please let us know what type of values you are supplying (e.g., daily maximums, monthly averages, annual averages). Additionally, if you have any reporting forms of other wastewater constituent records for the 1998, please attach them to this questionnaire. The energy consumed in any equipment used onsite to treat wastewater releases should be included in the utilities values on p. 4.
- 4. Concentration/units: Please specify the concentration of wastewater constituents generated in 1998. Please utilize the units of mass-per-volume (e.g., mg/liter, lb/gal).

5. Wastewater treatment/disposal (WW T/D) method: See the Wastewater Treatment/Disposal Methods Index on p. iii for method abbreviations.

Table 6a.	Annual	Units	DQI for Wastewater Annual Quantity ²	DQI for Wastewater
Wastewater Stream	Quantity ¹			Constituents ²

Table 6b.	CAS number	Concen- tration ⁴	Units	WW T/D Method	Table 6b (continued).	CAS number	Concen- tration ⁴	Units	WW T/D Method
Wastewater Constituents ³				5	Wastewater Constituents ³				5
Dissolved solids					Метсигу	7439-98-7			
Suspended solids					Lead	7439-92-1			
Chemical Oxygen Demand (COD)					Nitrogen	7727-37-9			
Biological Oxygen Demand (BOD)					Zinc	7440-66-6			
Oil & grease					Tin	7440-31-5			
Hydrochloric acid	7647-01-0				Ferrous sulfate	7720-78-7			
Sulfuric acid	7664-93-9				Ammonia	7664-41-7			
Other acids (please specify):					Nitrates				
1.					Pesticides				
2.					Other constituents:				
Phosphorus	7723-14-0				1.				
Phosphates					2.				
Sulfates					3.				
Fluorides					4.				
Cyanide	·				5.			÷	
Chloride					6.				
Chromium	7440-47-3				Wastewater comments:				
Iron	7439-89-6								
Aluminum	7429-90-5								
Nickel	7440-02-0								

7. HAZARDOUS & NONHAZARDOUS WASTES

1. <u>Hazardous wastes and EPA hazardous waste numbers</u>: Please list your waste streams that are considered hazardous by the U.S. EPA. Include the hazardous waste codes for any hazardous waste you include.

2. Annual quantity/units & Density/units: Please specify the amount of waste generated in 1998. Use units of mass-per-year (e.g., kg/yr, lb/yr). Please also provide the density for each waste.

3. Data quality indicators: See the Data Quality Indicators Index on p. iii for abbreviations. Please supply the DQI for the annual quantity value given.

4. Management method: See the Management Methods Index on p. iii for abbreviations. If none are applicable, please indicate other and use the comments section to expound.

5. <u>Transportation information</u>: See the Transportation Modes Index on p. iii for mode abbreviations. Please specify the one-way transportation distance, or the location to where the waste is shipped, and the number of trips made from your facility on an annual basis. % capacity represents what percent of the transport vehicle's total *load* was carrying the wastes of interest.

EPA Haz.	Annual	Units	Density ²	Units	DQI	Mgmt.	On or	Transportation Inf	ormatio	n (Shipp	ing) ⁵
Waste # ¹	Quantity ²				3	method4	offsite?	Dist. or Location	Mode	# trips	% сар.
F005	20,000	kg/yr	0.9	kg/liter	М	Ĭe	off	Indianapolis, IN	A	24	40
										1. ¹	-
	EPA Haz. Waste # ¹ FOOS	EPA Haz. Annual Waste # ¹ Quantity ² F005 20,000	EPA Haz. Annual Units Waste #1 Quantity2	EPA Haz. Annual Quantity ² Units Density ² F005 20,000 kg/yr 0.9	EPA Haz. Annual Quantity ² Units Density ² Units F005 20,000 kg/yr 0.9 kg/liter	EPA Haz. Annual Quantity ² Units Density ² Units DQI 3 F005 20,000 kg/yr 0.9 kg/liter M	EPA Haz. Annual Quantity2 Units Density2 Units DQI 3 Mgmt. method ⁴ Waste #1 Quantity2 0.9 kg/yr 0.9 kg/liter M. March 100 F005 20,000 kg/yr 0.9 kg/liter M. March 100	EPA Haz. Annual Quantity ² Units Density ² Units DQI 3 Mgmt. method ⁴ On or offsite? F005 20,000 kg/yr 0.9 kg/liter M. M. Momental offsite? F005 20,000 kg/yr 0.9 kg/liter M. M. Momental offsite? F005 20,000 kg/yr 0.9 kg/liter M. M. <t< td=""><td>EPA Haz. Annual Units Density² Units DQI Mgmt. method⁴ On or offsite? Transportation Info Dist. or Location F005 20,000 kg/yr 0.9 kg/liter M. M. Momt. method⁴ On or offsite? Dist. or Location F005 20,000 kg/yr 0.9 kg/liter M. M. Momt. method⁴ On or Transportation Info Dist. or Location F005 20,000 kg/yr 0.9 kg/liter M. M. Momt. method⁴ Off Site? Dist. or Location F005 20,000 kg/yr 0.9 kg/liter M. M.</td><td>EPA Haz. Annual Quantity² Units Density² Units DQI 1 Mgmt. method⁴ On or offsite? Transportation Information Waste #¹ Quantity² 0.9 kg/yr 0.9 kg/liter M Image: Arrow offsite? Dist. or Location Mode F005 20,000 kg/yr 0.9 kg/liter M Image: Arrow offsite? Image: Arrow offsite?</td><td>EPA Haz. Annual Units Density² Units DQI Mgmt. On or method⁴ Transportation Information (Shipp) Waste #¹ Quantity² 0.9 kg/yr 0.9 kg/liter M Image: Annual method 4 Dist. or Location Mode # trips F005 20,000 kg/yr 0.9 kg/liter M Image: Annual method 4 Image: Annual method 4</td></t<>	EPA Haz. Annual Units Density ² Units DQI Mgmt. method ⁴ On or offsite? Transportation Info Dist. or Location F005 20,000 kg/yr 0.9 kg/liter M. M. Momt. method ⁴ On or offsite? Dist. or Location F005 20,000 kg/yr 0.9 kg/liter M. M. Momt. method ⁴ On or Transportation Info Dist. or Location F005 20,000 kg/yr 0.9 kg/liter M. M. Momt. method ⁴ Off Site? Dist. or Location F005 20,000 kg/yr 0.9 kg/liter M. M.	EPA Haz. Annual Quantity ² Units Density ² Units DQI 1 Mgmt. method ⁴ On or offsite? Transportation Information Waste # ¹ Quantity ² 0.9 kg/yr 0.9 kg/liter M Image: Arrow offsite? Dist. or Location Mode F005 20,000 kg/yr 0.9 kg/liter M Image: Arrow offsite? Image: Arrow offsite?	EPA Haz. Annual Units Density ² Units DQI Mgmt. On or method ⁴ Transportation Information (Shipp) Waste # ¹ Quantity ² 0.9 kg/yr 0.9 kg/liter M Image: Annual method 4 Dist. or Location Mode # trips F005 20,000 kg/yr 0.9 kg/liter M Image: Annual method 4 Image: Annual method 4

Hazardous waste comments:

Table 7b.	Annual	Units	Density ²	Units	DQI	Mgmt.	On or	Transportation Inf	ormatio	n (Shipp	ing) ⁵
Nonhazardous Wastes	Quantity ²				3	method ⁴	offsite?	Dist. or Location	Mode	# trips	% сар.
EXAMPLE: Waste metal chips	22,000	kg/yr	1,000	kg/m3	С	R	off	225 km	A	2	100
1.											
2.											
3.											
4.											
5.											
6											
7.						3					
Nonhazardous waste comments:											

APPENDIX G

SUPPLEMENTAL MANUFACTURING DATA INFORMATION

this page intentionally left blank

APPENDIX G

SUPPLEMENTAL MANUFACTURING DATA INFORMATION

G-1. Frit Manufacturing Data Collection

No manufacturers were willing to supply complete frit manufacturing data as primary data for this study. Inventory data for the frit manufacturing process were obtained from secondary sources and from communications with industry representatives. Input data were derived from personal communication with an industry representative (Peer 2000) and output data were compiled from the U.S. Environmental Protection Agency's (EPA) *AP-42* publication (*Compilation of Air Pollutant Emission Factors*) (EPA 1997). The AP-42 data are provided for one ton of feed material, and it was assumed that the material efficiency is 100%, such that for each mass quantity of raw material input, the same mass quantity of frit is produced.

Limitations to these data are that a complete inventory, with all the information asked in the questionnaire for other manufacturing processes in this study, was not obtained. Personal communications were made to provide information on the major inputs to the process, based on a conversation taken place in June, 2000. The output data are based on a 1997 EPA publication. The publication date of the output data remains within the range of dates of primary data obtained in this study. Further, the frit manufacturing process is expected to be a relatively mature technology (compared to some LCD-related technologies), and 1997 data are expected to be representative of the monitors being studied in this project.

G-2. Printed Wiring Boards (PWBs) and Electronic Component Manufacturing Data Collection

Each display technology requires electronic printed wiring boards (PWBs) and their associated components such as integrated circuit (IC) chips, resistors, capacitors to operate the displays, independent of the computer's central processing unit. Therefore, the display PWBs and components are within the bounds of the analysis.

PWB and semiconductor (IC) manufacturing are highly energy and resource intensive processes. In purely a comparative analysis, we could consider eliminating the PWBs from the analysis since both the CRT and LCD display technologies use PWBs. However, the number of PWBs (4 major ones for LCDs and 2 for CRTs) and their makeup differ between the two technologies. For example, the AMLCD controller has more active parts because its addressing system is more complicated than the CRT. Therefore, exclusion of the PWB manufacturing process was chosen to be included in the scope of this project. In addition, beyond the goal of a comparative analysis between CRT and LCD, this study is intended to provide baseline data for each individual technology such that improvement assessments can be considered when evaluating the entire life cycle of a particular monitor. This provides another reason to include PWB manufacturing in the scope. However, due to the importance of collecting primary data for the other major display components (e.g., CRT tube and LCD panel/module manufacturing), a lower priority was given to obtaining PWB data. Given the lower priority for PWB and component data collection, questionnaires were not sent to multiple PWB and component manufacturers. Alternatively, data were obtained from an industry contact knowledgeable in PWB manufacturing (Sharp 2000). PWB component manufacturing data were not obtained; however, materials use as well as energy consumption from manufacturing PWB components are expected to be small in comparison to the overall manufacturing requirements for the CRT and LCD monitors; therefore, lack of PWB component manufacturing data is not expected to have a significant impact on the results.

REFERENCES

- EPA (Environmental Protection Agency). 1997. Compilation of Air Pollutant Emission Factors (AP-42). Fifth Edition, Volume I. June.
- Peer, J. 2000. Personal communication with J. Peer, Techneglas, and J. G. Overly, University of Tennessee, Center for Clean Products and Clean Technologies. June.
- Sharp, J. 2000. Personal communication with J. Sharp, Teredyne Corp.; J. G. Overly and M. Socolof, University of Tennessee, Center for Clean Products and Clean Technologies. August.

this page intentionally left blank

APPENDIX H

TECHNICAL MEMORANDUM: Use Life-Cycle Stage Approach

this page intentionally left blank

APPENDIX H

TECHNICAL MEMORANDUM: Use Life-Cycle Stage Approach

1. INTRODUCTION

1.1 Background

As part of the Environmental Protection Agency's Design for the Environment Program (DfE) Computer Display Project (CDP), the University of Tennessee Center for Clean Products and Clean Technologies is conducting an environmental life-cycle assessment (LCA) of 17" cathode ray tube (CRT) and 15" active matrix liquid crystal display (LCD) computer monitors. Chapter 1 of this report provides further details about the scope and boundaries of this project. As typically defined in LCA, the five main life-cycle stages of any product are as follows:

- Materials Extraction;
- Materials Processing;
- Product Manufacture;
- Product Use, Maintenance and Repair; and
- End-of-Life.

This technical memorandum (TM) presents the CDP's approach to developing the inventory that will be used to assess the environmental and health impacts from the use life-cycle stage of computer monitors. Maintenance and repair are not included within the boundaries of this analysis because they are expected to be minor contributors to environmental impacts compared to use and other life-cycle stages.

The purpose of this TM is to present the approach for developing the inventory of inputs and outputs associated with the actual use of the monitors. The final use-stage inventory will consist of electricity consumption from use as well as the indirect inputs and outputs from the generation of that electricity. The focus of this TM is on identifying the amount of electricity consumed during use. The inventory from electricity generation is presented in Appendix E. The final use-stage inventory combines these two sets of data and will be presented as part of the final LCI in Section 2.7 and Appendix J of this report.

In addition to energy consumption, other environmental or health issues have been associated with the use of computer monitors, including eye strain, ergonomics, and exposure to electric and magnetic fields. However, quantitative methods for assessing these impacts within the project LCA framework are not available. Thus, these impacts will be addressed qualitatively in the final LCA report for the project.

1.2 Calculating Energy Consumption During the Use Stage

CRTs and LCDs use different mechanisms to produce images on screen, which result in different energy use rates. These energy use rates can be combined with the number of hours a desktop monitor is on during its lifespan to calculate the total quantity of electrical energy consumed during the use life-cycle stage. In this project, two lifespan scenarios are considered:

- Manufactured life the amount of time either an entire monitor or a single component will last before reaching a point where the equipment no longer functions, independent of user choices.
- Effective life the actual amount of time a monitor is used, by one or multiple users, before it is disposed of, recycled, or re-manufactured. Reuse of a monitor by a subsequent user is considered part of its effective life. Recycling, on the other hand, is the reuse of parts or materials that require additional processing after disassembly and it is not considered part of the use stage.

These two scenarios are considered in this project in order to account for potential differences between how consumers *currently* use the equipment and how consumers could use the equipment. Currently, consumers often replace monitors before they physically break down. This behavior results in a lifespan that is not dependent on the monitor technology itself. The manufactured life, on the other hand, is based on the technology and represents how consumers could potentially use the equipment. If the lifespans are significantly different, the difference could have a large impact on how the use stage compares to the other life-cycle stages in this study. The remainder of this TM is broken down into the following sections: Methodology, Preliminary Results, Data Sources and Quality, Limitations and Uncertainties, and Discussion and Conclusions.

2. METHODOLOGY

As discussed in the previous section, calculating electrical energy consumption during the life of an electrical component requires two main pieces of information: the component's energy use rate (typically in watts or kilowatts) and the amount of time the component can or does spend in use (in hours per life).

Once energy use rates and hours per life in each mode are known, they can be multiplied to derive the total number of kilowatthours (kWh) consumed during the lifetime of a monitor according to the following general equation where "mode I" indicates the power consumption mode of a monitor (i.e., full-on or low, discussed further in Sect. 2.1). This basic equation will be used to calculate the total kWhs consumed over the manufactured and effective lives for LCDs and CRTs.

$$\sum_{i=1}^{2} \left[kW \text{ consumed in mode } i_{i} x \left(\frac{hours}{life} \text{ spent in mode } i \right) \right] = \frac{kWh}{life}$$

Section 2.1 presents information on the various energy use rates that exist for the project functional units in different power modes. Section 2.2 presents the methodology for calculating hours per life under the manufactured and effective life scenarios.

2.1 Energy Use Rate

Most desktop monitors manufactured today are built to use several different power consumption modes during normal operation. There are often up to four different power consumption modes that can be used by a monitor in going from a state of active use to a state of almost complete shut-down. These four modes, from greatest power consumption to least, are typically entitled 'full-on' or active use, 'standby,' 'suspend' and 'active-off.' For this TM, manufacturers' data on these power modes were collected from company contacts and Internet sites for 35 different 17" CRT monitors and 12 different 15" LCD monitors. The complete list of these data is presented in Attachment A, Table A1.

For the purposes of this TM, the power consumption modes have been categorized into two modes: 'full-on' and 'low.' The 'low' power mode is an average of the three low power modes typically provided by the manufacturers (i.e., standby, suspend, and active-off). These three categories were averaged to create one 'low' power consumption mode because hours per use data (needed for calculations in this TM) are only available for a 'full-on' and a reduced power mode. The low mode value for the CRT is the average of the three modal averages of standby, suspend and active-off. For the LCD, data on only two low-power modes (standby and active-off) were provided by manufacturers (see Attachment A, Table A2), and therefore, the low mode value is an average of those two modal averages. Table 1 presents the average values for full-on and low power modes that were used for subsequent calculations in this TM.

Monitor Type	Full-on Power Mode		Low Pow	er Mode ^b
	(W)	(kW)	(W)	(kW)
17" CRT	112	0.113	13.1	0.013
15" LCD	39.7	0.040	6.44	0.006

 Table 1. Average Energy Use Rates ^a

^a See Attachment A, Table A1 for source data.

^b An average of company-reported values for standby, suspend and active-off (see Attachment A, Table A1).

2.2 Calculating Lifespan

As stated previously, lifespan calculations in this TM are based on two different scenarios: manufactured life and effective life. Sects. 2.2.1 and 2.2.2 present the methodology and data needed to calculate energy use under each of these two scenarios, respectively. Sect. 2.2.2 is further divided into discussions of office versus home use patterns, the amount of time a monitor is operating in each power mode, and the number of years the monitor is operating in its lifetime. These results will be combined with data from the energy use rate calculations in Sect. 2.1 to obtain the energy consumption per life for each scenario and for each monitor.

2.2.1 Scenario #1: Manufactured Life

The manufactured life is defined here as the length of time a monitor is designed to operate effectively for the user. It is the number of hours a monitor would function as manufactured, and is independent of user choices or actions. One way to estimate this manufactured life is to use the mean-time-before-failure (MTBF) specification of a monitor or its components. The CRT MTBF specification dictates the amount of time the display must operate before it reaches its brightness 'half-life,' or the ability to produce 50% of its initial, maximum brightness. The MTBF value, generally provided in total hours per life of a monitor, is what most final manufacturers or assemblers of personal computer (PC) equipment, including monitor assemblers, typically specify for a component. To meet the specification, suppliers typically calculate the MTBF (a military-based specification) based on component data. Suppliers' test results are usually called the 'calculated' MTBF. The MTBF value also depends on which combination of power modes are used during testing, which is referred to as the 'duty cycle' and each supplier may use a different duty cycle to test their component.

Additionally, monitor assemblers will often perform their own testing, typically entitled 'demonstrated' MTBF. The testing includes sequences where the monitor is 'stressed' by quickly switching back and forth from an all black picture to an all white one, or quickly switching individual pixels either on and off or through multiple colors or black and white. Manufacturers typically find that their demonstrated MTBF is on the order of twice as long as the calculated MTBF (McConnaughey 1999, Douglas 1999). However, it should be noted that the demonstrated MTBF is not a real-time testing method, as the testing data is used in a complex equation to calculate that 'demonstrated' value.

From review of the information obtained on CRT-based monitors (see Attachment A, Table A2), it appears that the CRT itself is the limiting component, or the component that 99% of the time determines whether the entire monitor has reached its end-of-life. Thus, from the limited information that was obtained on CRTs, and the limited confidence that can be instilled in that data, an average of the two ranges obtained on the estimated lifetime of CRTs (10,000 - 15,000 hours) was used as the CRT manufactured lifetime (12,500 hours) (Goldwassar 1999, Douglas 1999).

For active matrix LCDs, the components that have the greatest potential to fail first are the display panel itself (including the liquid crystals and thin-film transistors), backlights, driver integrated circuit (IC) tabs, and other smaller components. The backlights and driver IC tabs can be field-replaced, thus their failure does not necessarily represent the end of the monitor's life. However, failure of the liquid crystals or transistors, which would require replacement of the display panel itself, would most likely mean that the monitor cannot be cost-effectively repaired. The MTBFs of all these components appear to have a broad range. For example, different backlight manufacturers reported from as few as 15,000 hours to as many as 50,000 hours (Douglas 1999, Tsuda 1999, VP150 1999). However, it appears that those components that are not field-replaceable (e.g., the LCD panel) have MTBFs in the range of 40,000 - 50,000 hours (Tsuda 1999, Young 1999). Thus in this TM, the amount of time an LCD monitor would operate during its manufactured life is assumed to be the average of the two non field-replaceable values, or 45,000 hours. In order for a monitor to operate for 45,000 hours, any major field-replaceable parts that have MTBFs less than 45,000 hours will need to accounted for in this LCA project. For example, assuming the backlights last on average 32,500 hours (the average of the values obtained for backlights), two would be needed for every panel during its lifetime. Therefore, in the final CDP LCA, the manufacturing of these type of components would need to be included in the inventory.

Little information is available on the duty cycles that component manufacturers use to test components. Thus, it is assumed that the average duty cycle utilized in testing components is 50% of the time tested in full-on mode and 50% in a lower power mode. Table 2 shows the values that are used in this TM for the hours per manufactured life for the CRT and LCD. The LCD manufactured life (45,000 hours) is 3.6 times greater than the CRT manufactured life (12,500 hours). Therefore, based on equivalent use periods, 3.6 CRTs would need to be manufactured for every single LCD.

Monitor Type	Total Hours (hours/life)	Mode	Duty Cycle (% time spent in each mode during testing)	Hours per Mode (hours/life)
17" CRT	12,500	Full-on	50%	6,250
		Low	50%	6,250
15" LCD	45,000	Full-on	50%	22,500
		Low	50%	22,500

 Table 2. Manufactured Life Values

2.2.2 Scenario #2: Effective Life

The effective life scenario attempts to model the actual quantity of hours that an average monitor spends in each of the two primary power consumption modes (full-on and a lower power state) during its lifetime. The effective life of an average monitor is based on the following information:

- The proportion of computers that are used in an office environment versus a home environment, to account for different use rates in these two basic user environments;
- The amount of time in a year a typical monitor spends in full-on power mode and in a lower power-consuming mode for both office and home environments; and
- The number of years a typical monitor is used during its lifespan for both office and home environments, not including years in storage before a monitor is replaced or discarded (as it is not consuming power during storage).

Under this lifespan scenario, we assume there is no difference in the amount of time a CRT or LCD monitor is operating. That is, the hours per life for the effective life calculation is not technology-dependent. Therefore, the same set of hours-per-life values are used to calculate the kWhs used per effective lifetime for a CRT and an LCD. The remainder of this section discusses the data and methods used to calculate the final hours-per-life values used in the effective life scenario. In order to obtain these final values, we need to determine the percentage of office versus home environment users, the annual use operating patterns in the office and home environments (hours/year) within each power mode, and the number of years a monitor is in operation during life. The following three subsections address these data needs.

2.2.2.1 Percentages of Office- and Home-Environment Users

Home and office users of computer equipment do not follow the same use patterns. Thus, data are needed on the percent of users in each environment to determine the use pattern of an "average" computer monitor. The most recent data available for both home and office users are for 1997. The Computer Industry Almanac for 1997 reports an estimated 117 million total computers were in use in the United States in 1997 (CIA 1997). In addition, the 1997 Residential Energy Consumption Survey (RECS) reports that 43 million PCS were used in homes in 1997 (EIA 1999). Therefore, assuming the remaining non-household computer monitors are all in office environments, there would be approximately 74 million computers being used in office environments.

Note that an 'office' environment may be a school, hospital, or other commercial environment, and the computers they use may follow widely varying degrees of use. For example, computers (and thus monitors) in a school may only be used a few hours in a day, while hospitals might operate theirs nearly constantly. For this TM, it is assumed that on average, typical office use patterns (to be presented in Sect. 2.2.2.2) are representative of all non-home environment users.

The 1997 RECS also reported that 6% of the 43 million household computers were used to telecommute (EIA 1999), which equals approximately 2.6 million computers. The use pattern of a telecommuter is assumed to resemble more closely an office environment than a home environment; therefore, the number of office environment monitors is assumed to total 76.6 million. Therefore, for purposes of calculations in this TM, the percentage breakdown of office and home environment monitors in the United States is as follows:

- Office: (74 million + 2.6 million) / 117 million = 65%
- Home: (43 million 2.6 million) / 117 million = 35%.

2.2.2.2 Operating Pattern (Average Hours in Use Per Year)

In order to determine the amount of electricity consumed during a monitor's effective life, we need to know the use operating patterns for both the office and home environments. The 'operating pattern' is defined here as the number of hours per year spent in each power mode. The average number of hours per mode per year will be the weighted average of the two user environments (i.e., 65% office, 35% home).

A literature search for computer monitor operating patterns was conducted for both office and home environments and a summary of literature reviewed is presented in Attachment A, Table A3. For data on office environment operating patterns, the most relevant and complete information found was from work performed by Lawrence Berkeley National Laboratory (LBNL) presented in a report entitled "Measured Energy Savings and Performance of Power-Managed Personal Computers and Monitors" (Nordman et al. 1996). Their definition of the standard operating pattern was based on earlier work performed by LBNL that studied electricity use by office equipment in commercial buildings, and referenced multiple studies on the use of office equipment, some having sample sizes as large as several hundred systems. Table 3 presents the standard office operating pattern for three different types of days (workday, weekend day and absence day), based on Nordman et al. (1996). Note that Nordman et al. "...first distinguish between weekdays and weekend days, with the latter including only Saturdays and Sundays. Then, any weekday which has less than half an hour of on-time (full-on or low power) is considered an absence day; the rest of the weekdays are workdays." Therefore, absence days may include some hours in operation. Also, some individuals may leave their computers on while out of the office, also resulting in hours in operation while the user is out of the office.

Type of Day	Standard Office Operating Pattern						
		On					
	Full-on	Low	Total				
Workday (hr/day)	4.1	8.4	12.5	11.5			
Weekend Day (hr/day)	0.0	4.8	4.8	19.2			
Absence Day (hr/day)	0.0	4.8	4.8	19.2			
Average Day ^b (hr/day)	2.3	6.9	9.2	14.8			

 Table 3. Standard Office Operating Pattern^a

Source: Adapted from Nordman et al. 1996, based on percentage of time in each mode.

^a Based on the assumption that all monitors take advantage of low power modes.

^b To calculate the average day, an average week is assumed to consist of 4 workdays, 2 weekend days, and 1 absence day per week. Average monitor usage per day weighs each average day by the number of each day type in a

week.

APPENDIX H

Using the hours per day for the three basic day types, we created an average annual day for the purposes of eventually obtaining an annual average. To calculate the average day, we assumed that a typical week of computer use in the year contains 4 workdays, 2 weekend days, and 1 absence day. By including 1 absence day in the typical week, the calculation results in 52 days annually that are days of no active computer system use, and are intended to include the following: holidays, sick days, vacations, work travel and days when computers are not needed for work or are not in use in the office.

LBNL's operating patterns do not appear to take into account whether or not a monitor is actually taking advantage of the various power savings modes. In their 1996 study, Nordman et al. found that only one-third of the monitors were set up to recognize time-based indicators that power down a part of the monitor or part of the PC that sends information to the monitor. More recently, a representative of the EPA Energy Star Program estimated that approximately 90 -95% of those monitors manufactured and sold in tandem with a PC in 1998 were pre-set up to take advantage of the multiple power consumption modes of the monitor, without any setup by the user (Fanara 1999). However, monitors are also sold individually, and no statistics were found on how many are sold that way and what percent of those are able to work with a PC's energy savings systems without assistance from the user. For the purposes of this TM, we assume that 90% of the monitors manufactured in 1998 and in use today are set up to recognize the power management signaling either from the monitor itself or from the PC to which they are connected. The lower end of the 90 - 95% range was chosen to recognize that monitors sold separately were not accounted for, and also, from an environmental impact perspective, it is conservative to assume a lower percent, which means less use of power saving features and greater energy consumption.

It should also be noted that this 90/10 split takes several assumptions into account, including but not limited to the percentage of users who alter or change their PC's and/or monitor's energy-saving settings, the percentage of users who know how to alter or change their PC's and/or monitor's energy-savings settings and the number of small-sized companies that build PC systems and whether or not they configure their systems to be able to take advantage of energy-savings settings 'out-of-the-box.' Because our confidence in this percentage split is not high, we will perform a sensitivity analysis of different percentage breakdowns of using low power modes versus not using them (50/50, 75/25, and 100/0; presented in Sect. 3.3).

This 90/10 split of using versus not using the power saving modes is implemented in the calculations by adjusting the average amount of hours per day a monitor spends in each mode for the effective life calculations (see Table 3). Thus, 10% of the value of each number in the 'Low' column of Table 3 was removed and added to the 'Full-on' value in that same row, to account for those that cannot go into a lower power-saving mode. Table 4 presents the adjusted figures for hours per day and presents the annual average values by multiplying the average day values by 365 days. The average day is calculated in Table 4 the same way it was calculated in Table 3.

Type of Day		Adjusted Office Operating Pattern						
		On						
	Full-on	Low	Total					
Workday (hr/day)	4.9	7.6	12.5	11.5				
Weekend Day (hr/day)	0.48	4.3	4.8	19.2				
Absence Day (hr/day)	0.48	4.3	4.8	19.2				
Average Day ^b (hr/day)	3.0	6.2	9.2	14.8				
Annual Average (hr/yr)	1,095	2,263	3,358	5,402				
Percent on time spent in each mode	33%	67%	100%					

Table 4. Adjusted Office Operating Pattern^a

^a Values in Table 3 have been adjusted based on the assumption that 90% of monitors can take advantage of low power modes. Therefore, 10% of the hours in low mode in Table 4 were added to the full-on column and subtracted from the low column in this table.

^b To calculate the average day, an average week is assumed to consist of 4 workdays, 2 weekend days, and 1 absence day per week. Average monitor usage per day weighs each average day by the number of each day type in a week.

For the home environment operating patterns, the most relevant and complete information was found in the RECS report (EIA 1999). The survey contained data on the use of computers in the home and how many hours per week the users have their computer on, without distinguishing power mode. Table 5 reveals the information obtained from the RECS report and breaks that data down to calculate a daily average and then an annual average operating pattern (i.e., the total number of hours of on time in one year).

Use Frequency Category	U.S. Households v (EIA 1	vith Computers 999)	Average Hours in Use for Each Category	Average Household Use ^a	
	(millions of households)	(% of households)	(hours/week)	(hours/week)	
Less than 2 hours per week	8.2	23.0%	1	0.2	
2 to 15 hours per week	17.4	48.9%	8.5	4.2	
16 to 40 hours per week	6.7	18.8%	28	5.3	
On all the time	3.3	9.3%	168	15.6	
Totals	35.6	100.0%		25.2	
Daily Average (hours per day)				3.6	
Annual Average (hours per year)			1,315	

 Table 5. RECS Home Operating Pattern Breakdown

Note: Totals may not be additive due to independent rounding.

^a These values are the product of the fraction of households in each category and the average hours per week in each category.

Data on the amount of time a home-environment monitor is in full-on versus a lower power mode was not provided in the RECS, nor was such data found elsewhere. Thus, lacking any other information, we have chosen to use the percentage breakdown found in the office-environment data for the home-environment data in this TM (see the bottom row of Table 4). These percentages are applied to the total 1,315 hours/year for home-environment use to estimate the amount of time in each mode. In addition, the 90/10 split of equipment that can/cannot go into lower power saving modes was applied to these values to determine the actual expected number of hours per year per mode for the home environment. In Table 6, the hours per year values for each power mode are shown. In order to determine these hours spent in each mode, the total number of hours spent on annually (1,315 hours/year) was first split by the 90/10 factor into equipment that can and cannot save energy categories. Then, the resulting 1,183 hours/year was split by the office-environment data on the percent of time spent in each mode, resulting in 390 hours annually in full-on mode and 793 hours annually in a lower power mode. The remaining 132 hours/year that cannot go into an energy saving mode, was included in the 'Full-on' row. The two values for each row are then added to obtain the total hours annually in a home environment that a monitor would spend in each power mode.

Lastly, Table 7 shows the final values obtained for the effective life calculations, as presented in this section, for hours per year per mode for office- and home-environment users.

		8					
Power Percent of Time in Mode Each Mode		Time Operating in Each Mode (hours/year)					
	(from office- environment data)	90% That Can Save Energy	10% That Cannot Save Energy	Total			
Full-on	33%	390	132	522			
Low	67%	793	0	793			
Total	100%	1,183	132	1,315			

 Table 6. Splitting the Home Operating Pattern Data into the Two Power Modes

	Summary of Operating I	atterns for Effective Eff					
User	Time Operating in Each Environment						
Environment		(hours/year)					
	Full-On Power Mode	Low Power Mode	Total				
Office	1,095	2,263	3,358				
Home	522	793	1,315				

 Table 7. Summary of Operating Patterns for Effective Life Calculations

2.2.2.3 Average Years Per Life

The third set of values required for the calculation of hours per effective life is the number of years of use in the life of a monitor. The number of years per effective life, multiplied by the operating patterns in hours per year (presented in Table 7), will result in the hours per effective life.

A monitor may be reused in multiple 'lives' before reaching its end-of-life. The end-of-life is defined as the point at which the monitor is no longer used for its intended purpose in the physical form in which it was originally manufactured. End-of-life options include indefinite storage (in which case it is not reused after storage), de-manufacturing, recycling, or disposal. A monitor may be stored before being reused; however, this storage time will not affect our use calculations since no electricity is required to operate the monitor during this storage. After its first life as used by the original owner, a monitor might be used by different people and with different PC systems in subsequent lives.

For data on the number of years of use that are in a monitor's lifetime, several sources of information were reviewed. Two particular studies provided relevant data on the number of years per life (Matthews et al. 1997, NSC 1999). A study by Matthews et al. (1997), which was an update to a study originally performed in 1991, concluded that after a first life of 5 years, approximately 45% of all PC systems are reused, while the remaining 55% either go directly to recycling or landfilling (10%) or are stored and then recycled or landfilled (45%). Their study only addressed PC systems as a whole and did not break down the lifetimes of individual components. Additionally, they concluded that the period over which the systems are reused is 3 years.

In a recently completed study for the National Safety Council (NSC 1999), researchers interviewed more than 30 major manufacturers and resellers of CRT computer monitors and other computer components. NSC found that a CRT monitor's first life lasts approximately 4 years, while the total lifespan is on the order of 6 - 7 years. Since the NSC study contains results that pertain specifically to monitors, and provides the most recent data, its results are used in this TM. The values that are used for calculations in this section are 4 years for the first life of use, and 2.5 years for the second and subsequent lives of use. The operating pattern for monitors in all the years over its effective life (6.5 years) are assumed to be the same as presented in Sect. 2.2.2.2 (Table 7). However, in the lives subsequent to the first life, the hours per year values are reduced by the fraction of monitors assumed to be reused. Matthews et al. (1997) estimated that 45% of PCS are reused after a first life; thus, the effective life operating pattern values in years of life after the first life are 45% of the values in the first life (which were presented in Table 7).

While the NSC data singled out CRT monitors in their lifespan estimates, they did not single out desktop LCD monitors. Their data did contain estimates of a 'Notebook PC,' which were 2 - 3 years for the first life and 1 - 2 years for the remaining lives, however, we expect that desktop LCD monitors will more closely mirror the lifetime estimates of a desktop CRT monitor than that of a notebook PC. Consequently, it was assumed that LCD desktop monitors also spend 4 years in their first life and 2.5 years in their subsequent lives. Additionally, the NSC document did not attempt to separate those computer systems or monitors that are used in an office versus a home environment. Thus, it was assumed that the same years per life are realized for office and home environments.

2.2.2.4 Summary of Effective Life Values (Hours per Life)

Data presented throughout Sect. 2.2.2 that are needed to estimate the hours per effective life, are shown in Table 8. The values for hours per year per power mode, calculated in Sect. 2.2.2.2 and presented in Table 7 are assumed to be the operating pattern throughout the first life (first four years). In the remaining lives, the annual operating hours decreases to 45% of the hours in operation during each year in the first life, with the remaining lives lasting a total of 2.5 years (see Sect. 2.2.2.3). Table 8 also presents the total hours per effective life per mode, based on percentage in office and home environments. These values are in bold in Table 8 (4,586 and 8,961 hrs per effective life) and will be used with the energy use rates per mode (presented in Sect. 2.1, Table 1), to calculate the total energy consumption per effective life for each monitor type.

User Environment	Power Mode	Power First Life Mode (4 years)		Remaini (2.5 y	Model Totals ^b (hr/effective	
		Operating Pattern (hr/yr)	Total (hr/4 yrs)	Operating Pattern (hr/yr)ª	Total (hrs/2.5 yrs)	life)
OFFICE	Full-on	1,095	4,380	493	1,233	5,613
(65%)	Low	2,263	9,052	1,018	2,545	11,597
HOME	Full-on	522	2,088	235	588	2,676
(35%)	Low	793	3,172	357	893	4,065
WEIGHTED	Full-on					4,585
AVERAGE ^c	Low					8,961

 Table 8. Effective Life Values

^a The remaining lives operating pattern is 45% of first life operating pattern, based on 45% of monitors that are reused (Matthews et al. 1997).

^b Modal totals calculated as [(Total for first 4 years) + (Total for remaining 2.5 years)].

^c The weighted averages shown for full-on and low power modes are based on the assumption that 65% of users operate in an office environment and 35% operate in a home environment.

3. PRELIMINARY RESULTS

In order to calculate the total kWhs consumed per manufactured life and effective life, values from Sects. 2.1 and 2.2 were combined as shown in Tables 9A and 9B. First, the energy use rates (kW) were multiplied by the lifespans (hours per life) for each mode and each monitor type. They were then summed for the two power modes to obtain a total kWh/life for each monitor type. In an LCA, comparisons are made based on functional equivalency. Therefore, if one monitor will operate for a longer period of time than another, as in the manufactured life scenario, overall life-cycle impacts should be based on an equivalent use. Thus, because the manufactured life of an LCD is 3.6 times greater than a CRT (see Sect. 2.2.1), in the final analysis, the CRT manufacturing process inventories must be multiplied by 3.6 to retain a functionally equivalent basis for the CRT and LCD monitor comparison. Since the effective life calculation is not technology-dependent, both monitor types operate for the same number of

hours in the effective life (see Table 8) and thus they are considered functionally equivalent and no modification to the overall life-cycle analysis is necessary.

Monitor Type	Power Mode	Energy Use Rate (kW)	ML Calculated Lifespan (hours/life)	ML Energy Consumption (kWh/life)
17" CRT	Full-on	0.113	6,250	706
	Low	0.013	6,250	81
	Total		12,500	787
15"LCD	Full-on	0.040	22,500	900
	Low	0.006	22,500	135
	Total		45,000	1,035

Table 9A. Manufactured Life (ML) Electricity Consumption

Table 9B.	Effective	Life (EL)	Electricity	Consump	tion
-----------	-----------	-----------	-------------	---------	------

Monitor Type	Power Mode	Energy Use Rate (kW)	EL Calculated Lifespan (hours/life)	EL Energy Consumption (kWh/life)
17" CRT	Full-on	0.113	4,585	518
	Low	0.013	8,961	116
	Total		13,547	634
15"LCD	Full-on	0.040	4,585	183
	Low	0.006	8,961	54
	Total		13,547	237

3.1 Comparing Lifespans: Manufactured Life to Effective Life

Since the energy use rates are the same across both lifespan scenarios for CRTs and LCDs, we can compare the calculated lifespans of the manufactured and effective lives (hours per life). For the CRT monitor, the manufactured life total hours are 12,500 versus the effective life total of 13,547. While this does seem to suggest that a CRT can be used longer than is physically possible, what this brings out is the lower confidence we have in these numbers and some of their supporting values, with less confidence in the manufactured life data. Assumptions were required several times that could bias these numbers in either direction, however it is thought that most likely the manufactured life estimate is low based on the other estimates for the overall CRT monitor (see Attachment A, Table A2). However, there was no sound basis for assuming a lower value and thus the above hours per life values were used. It should also be stated that while these numbers are different, they are within an 8% error range of one another, and can be taken to be a near 1:1 ratio, indicating a similar potential lifespan.

For LCDs, the comparison across lifespan scenarios looks more like what one would expect, with the manufactured life value of 45,000 hours per life being much greater than the effective life value of 13,547 hours per life. The effective life value reflects the assumption that a user's use habits are not technology-dependent, and would seem to reveal that LCDs are not being used as long as they can physically be (less than a third as long).

APPENDIX H

The difference between the manufactured and effective lives are important when evaluating all the life-cycle stages for a particular monitor type. If the manufactured life is significantly greater than the effective life, the use stage will have greater impacts, as compared to other life-cycle stages. Therefore, it is important to focus on the lifetime scenario that is most realistic, while still recognizing the potential impacts from another feasible lifespan scenario.

In the final LCA for this project, we will use the effective life as the primary basis for the use stage inventory due to the fact that the effective life data are attempting to obtain a more realistic value for kWhs consumed per lifetime, and that we currently have greater confidence in those data versus the manufactured life data. The manufactured life data will be used to discuss potential differences in the use stage impacts based on this alternative lifetime scenario.

3.2 Sensitivity Analysis

Finally, in an effort to provide some sensitivity analysis to the final values, the assumption used in the effective life calculation that 10% of the computers manufactured in 1998 and currently in use are not able to take advantage of lower power-saving modes (a 90/10 split) was adjusted to three different splits, with all the other assumptions and calculations kept unchanged (50/50, 75/25 and 100/0, respectively in each case those that are able to go into power saving modes and those that are not). Table 10 presents the results of the sensitivity analysis for each of the four power-saving functionality scenarios.

Monitor Type	Power Mode		% that Can / % that Cannot Take Advantage of Power-Saving Features				
			50/50	75/25	90/10	100/0	
17" CRT	Full-on	(kWh/life)	969	689	518	408	
	Low		64	97	116	129	
	Total		1,033	786	634	537	
15" LCD	Full-on	(kWh/life)	343	244	183	145	
	Low		30	45	54	60	
	Total		373	289	237	205	

Table 10. Sensitivity Analysis of Effective Life Results

The data in Table 10 reveal that the final electrical energy consumption values for the CRT would increase by 63% with a 50/50 split and decrease by 15% with a 100/0 split (from the 90/10 split assumption). Similarly for the LCD, the results would increase by 57% or decrease by 14%. Varying the use of power-saving features results in variations in the total amount of energy consumed for LCDs and CRTs, but does not vary the ratio of LCD to CRT energy use. Therefore, these variations will affect the magnitude of the use stage impacts for effective life scenarios when compared to other life-cycle stages, but will not affect the comparison of LCD to CRT. Additional sensitivity analyses are available in Socolof et al. (2000).
4. DATA SOURCES AND QUALITY

Source and quality information for the data utilized in this TM are detailed in Table 11. Four categories of data quality ratings were assigned: excellent, average, poor, and unknown. In general, data assigned higher quality ratings were directly measured and represent 1998 data. As data required more calculation or estimation, or were found from a previous year, the data quality rating was reduced.

In general, the overall level of data quality is between average and excellent. However, a distinct difference can be seen in the average data quality ratings given to manufactured life data (average) and the effective life data (excellent). This infers that greater confidence can be placed in the effective life data than in the manufactured life data. Additionally, the energy use rate data appears to be of average.

Data	Data Source/References	Data Source Comments	Data Quality ^a	Data Quality Explanation
ENERGY USE RATE				
Various monitor manufacturer's energy use rate data	Web sites in most cases; E-mail from manufacturer in remaining cases.	It was assumed that the data provided by manufacturers on the Web sites were high- quality data in that the data should be measured and directly applicable to the equipment for which the information is provided. However, the search for information did not separate information obtained by performance level of the monitors.	Average	It was not possible to determine in what year each individual monitor was manufactured; however, it is assumed that each monitor is on the order of several months to 2 years old when promoted for sale. Thus it is estimated that the average date of the information obtained is probably relative to approximately 1997. Adding that the data was not sorted by performance level, this data was given a data quality rating of Average.
MANUFACTURED LIF	E (Only those sources	utilized to derive values are discussed here	.)	
Discussion of CRT lifespan	McConnaughey 1999	Professional opinion provides good insights into potential ranges for certain components, however is still an opinion and not scientific data.	Average	As a computer manufacturing company employee, it is expected that they are a quality source of information on this topic; however, information is still an opinion and not scientific data, thus an Average data quality rating was assigned.
Discussion of CRT and LCD lifespans	Douglas 1999	See above comment.	Average	See above comment.
Discussion of LCD lifespan	Ritsko 1999	See above comment.	Average	See above comment.
Discussion of LCD lifespan	Tsuda 1999	See above comment.	Average	See above comment.
Discussion of CRT and LCD lifespans	Young 1999	See above comment.	Average	As the leader of a group that closely follows the trends in the LCD market and produces monthly reports on technology and market trends, it is expected that they are a quality source of information on this topic, however, information is still an opinion and not scientific data, thus an Average data quality rating was used.
17" CRT monitor specifications sheet	VP150 1998	As technical data on one specific CRT monitor the information is expected to be at least testing quality data or better.	Excellent	As direct manufacturer information applicable to 1998, this data is given an Excellent data quality rating.

Table 11.	Data Sources and (Duality	v Information	for the	Use I	ife-Snan	Stage TM
	Data Sources and v	Zuani	y mior mation	ior the		me-span	Stage Int

Data Data Source/References		Data Source Comments	Data Quality ^a	Data Quality Explanation	
EFFECTIVE LIFE					
Number of PCS in use in CIA 1997 the U.S.		Authors have much experience in obtaining and collecting computer statistics in U.S. and other countries; have been publishing this book since 1986.	Excellent	From review of the available information on the authors and data sources for the data that go into the Computer Industry Almanac, the data quality rating of Excellent is given. Even though data is from 1996, the authors used that data and recent trends information to predict 1997 values, and it is expected that the 1997 values are not significantly different than the 1998 values.	
Percent of PCS in the home that are used in an office-like environment	nt of PCS in the that are used in an -like environment RECS 1999 "The Residential En- Survey provides nar about U.S. househo usage. The 1997 su from a statistically s 5,902 households th their homes."		Excellent	While these data are 1997 data, it is assumed that the energy usage patterns of home dwellers has not changed significantly between 1997 and 1998.	
Number of PCs in use in the home	RECS 1999	See previous comment on RECS.	Excellent	See previous comment on RECS.	
Office PC use pattern	Nordman 1996	Used multiple sources of previous data covering many samples of PCs, as well as their own research, to derive their equipment usage pattern.	Average	Their data was manipulated slightly to account for a greater number of affectors on typical usage patterns. By manipulating their data the data quality is slightly reduced, thus data quality rating of Average was assigned.	
Home PC use pattern	RECS 1999	See previous comment on RECS.	Excellent	See previous comment on RECS.	
Number of years PCS are used in 1st life and 2nd and subsequent lives	NSC 1999	"This study presents the results of the first large-scale survey (which covered the years 1997 and 1998) and analysis of end- of-life electronic product recycling and reuse in the U.S. Data were collected from 123 firms."	Excellent	Due to the applicable time frame and the body of companies who participated, these data were given data quality rating of Excellent.	
Number of PCS that are used in their 2nd and subsequent lives	Matthews 1997	Performed a study in 1991, watched as the computer market changed over 16 years, then reviewed the original study finding the weak spots there. Reperformed study in 1997 making learned changes to the analysis format and using newer data (1997).	Average	While changes were made to the second study as weaker parts of previous study were uncovered, still extrapolated individual recycling firm data to obtain some base data for their estimates. While data is primarily relative to the 1997 time frame, which is very close to our year of interest of 1998, still chose data quality rating of Average due to amount of data manipulation that was required to obtain values.	

Table 11.	Data Source	s and Ouali	tv Information	for the Use	e Life-Span	Stage TM
I WOIV III	Data Source	s and y and	y intor matter			

^a The data were assigned to one of the following four data quality categories: Excellent, Average, Poor, and Unknown.

APPENDIX H

5. LIMITATIONS & UNCERTAINTIES

This section is subdivided into three subsections, with each addressing the limitations and uncertainties of the energy use rate, manufactured life and effective life calculations.

5.1 Energy Use Rate

The energy use rates utilized in this TM were not from a systematic study of the energy use rates of all applicable monitors, only those for which information was located on the World Wide Web. Also, it was difficult to pinpoint the exact date from which much of the data came (see the data quality explanation of those data in Table 11). To successfully and effectively take advantage of the data on each of the three lower power modes, it would have been necessary to have hours per life data for each of the three power modes as well as for manufactured and effective lives. No sources of data, for either lifespan, separated hours per life estimates into three distinct low power modes. While this does induce error, it is expected that averaging those categories to estimate the total amount of time that a monitor spends in all the lower power modes would only have a minor effect on the final energy use rate values. The effect of averaging these categories probably overestimates the total amount of electrical energy that is consumed during lower-power mode use. This is so because those that are left on for significant periods of time (overnight, over a two-day weekend, or over an extended stay) most likely are reaching their lowest power mode within the first 1 - 2 hours and staying there for the duration of the away time.

When this information was obtained from the World Wide Web, the data were simply separated into one of the two large categories of 17" CRT and 15" LCD desktop monitors. Thus, since it is fairly common that one type of monitor manufacturer will make several different models with varying performance characteristics with one size range, a limitation of this data is that it is not sorted by performance characteristics. Additionally, the data obtained from these Web sites are most likely maximums, and were stated as such in several cases. However, if some manufacturers did not state that the reported values were maximums, then our averages are slightly high.

5.2 Manufactured Life

Only a very limited amount of information was obtained with which to make the assumptions made in this TM about manufactured life. The primary uncertainties relate not only to the assumption of the MTBF lifespan of the monitors, but also to the testing duty cycle which was completely estimated. With the lack of any high quality data, the confidence in the manufactured life calculations is low.

5.3 Effective Life

Several assumptions were made to calculate the effective life data set. They include the following:

- 90% of monitors are able to go into lower power consumption modes;
- Atypical workplace computers (e.g., those used in hospitals and schools) balance to fit the office-like use environment established in this TM;
- Atypical (average) office use environment week consists of 4 'weekdays,' 1 'absence day' and 2 'weekend days,' where the absence day accounts for holidays, sick days, work-related travel days, vacations and days of no use of the work computer;
- The percent of time a monitor spends in full-on versus lower power modes in a home environment is the same as in an office environment;
- Used the office environment split of the total on time that equipment spends in full-on versus a lower power mode for the home environment split of total on time;
- LCD desktop monitor lifetimes are more similar to desktop CRT monitors than notebook PC displays; and
- The same number of years of use per life exist for office and home environments.

While the above assumptions do introduce error, the magnitude of the error is unknown. Some assumptions may have a greater effect on the final values than others. For example, it may be concluded that the assumption that the average office PC system usage pattern is fairly accurate, while the assumption about atypical workplace computers could potentially contain significant error in either direction of the assumed value. Table 10 showed the effects on the results from varying the use of energy saving features. If several of these assumptions are biased in the same direction (either all underestimating or overestimating the results), then the effective life results have the potential to be significantly under or overestimated.

6. DISCUSSION AND CONCLUSIONS

The information presented in this TM are used to calculate the environmental burdens generated during the use life-cycle stage of the monitors. That information is then compared to those burdens that occur in the other four life-cycle stages of materials extraction, materials processing, manufacturing and end-of-life. To calculate the environmental burdens from the use life-cycle stage, the results of this TM -- the values for each monitor's electrical energy consumption over its lifetime in kWhs -- are multiplied by each of the inputs and outputs from the electricity generation process.

APPENDIX H

ACRONYMS & ABBREVIATIONS

CDP	Computer Display Project
CRT	Cathode ray tube
DfE	Design for the Environment
DOE	Department of Energy
DPMS	Display Power Management Signaling
EIA	Energy Information Administration
IC	Integrated circuit
kW	Kilowatt
kWh	Kilowatthour
LBNL	Lawrence Berkeley National Laboratory
LCA	Life-cycle assessment
LCD	Liquid crystal display
LCI	Life-cycle inventory
MTBF	Mean-time-before-failure
PC	Personal computer
RECS	Residential Energy Consumption Survey
TM	Technical memorandum
W	Watt

REFERENCES

Atlantic Consulting. 1998. "LCA Study, EU Ecolabels for Personal Computers." January.

- CIA (Computer Industry Almanac, Inc.). 1997. Computer Industry Almanac. Arlington Heights, IL.
- CCPCT (Center for Clean Products and Clean Technologies). 1999. The Calculated Averages of Energy Use Data Found from Multiple Computer Manufacturers' Web Sites.
- Chan, A., *et al.* 1997. "The Role of Computers in Energy Consumption on Campus." A class research paper from The University of Toronto. Available from the Web site: http://server.esc.cquest.utoronto.ca/env/env421h/energy/computers.html.
- Douglas, D. 1999. Personal communication with D. Douglas and E. Korman, Dell, and J. Overly, University of Tennessee, Center for Clean Products and Clean Technologies. March.
- EIA (Energy Information Administration). 1999. 1997 Residential Energy Consumption Survey, Housing Characteristics report. Web site available at: http://www.eia.doe.gov/emeu/consumption/reports.html (under "Residential Households").
- EPA (Environmental Protection Agency). 1999. Energy Star-Labeled Office Equipment with Computer Specifications. Web site available at: http://www.epa.gov/appdstar/esoe/pcsum.html.
- Fanara. 1999. Personal communication with Fanara, Environmental Protection Agency (EPA) Energy Star Program, and J.G. Overly, University of Tennessee, Center for Clean Products and Clean Technologies. July.
- Goldberg, L. 1998. "The Advent of "Green" Computer Design." Computer Magazine. September.
- Goldwasser, S.M. 1999. "Notes on the Troubleshooting and Repair of Computer and Video Monitors." Web site available at: http://plop.phys.cwru.edu/repairfaq/REPAIR/F_monfaq.html.
- Jung, L.B. 1999. "The Conundrum of Computer Recycling." Resource Recycling. May.
- Koch, J. 1996. "Evaluating Environmental Performance: A Case Study in the Flat Panel Display Industry." Submitted in partial fulfillment of the requirements for the degree of Master of Science at the University of Michigan. April.

- Maginnovision. 1998. Information found on a Web page from Maginnovision. Web site available at: http://www.maginnovision.com.
- Matthews, H. S., *et al.* 1997. Disposition and End-of-Life Options for Personal Computers. Green Design Initiative Technical Report #97-10, Carnegie Mellon University, Pittsburgh, PA. July.
- McConnaughey, M. 1999. Personal communication with M. McConnaughey, ViewSconic, and J. Overly, University of Tennessee, Center for Clean Products and Clean Technologies. March.
- Miseli, J. 1999. "Hidden Cost Savings of Flat Panel Displays." Display Manufacturing Technology Conference Digest of Technical Papers, San Jose, CA. February.
- NSC (National Safety Council). 1999. Electronic Product Recovery and Recycling Baseline Report: Recycling of Selected Electronic Products in the United States, Washington, D.C. May.
- Nordman, Bruce, M.A. Piette and K. Kinney. 1996. Measured Energy Savings and Performance of Power-Managed Personal Computers and Monitors. Presented at the American Council for an Energy Efficient Economy Summer Study, Pacific Grove, CA, LBL-38057, UC-1600. August.
- Philips. 1998. c1995 Typhon 19" high resolution digital autoscan monitor (CRT) fact page. Web site available at: http://www.fimi.philips.com/product/graphic/c1995.html.
- Planetmac. 1999. Information found on a Web page from PlanetMac. Web site available at: http://planetmac.com/Frames/catalog/monitors/gdm1962.html.
- Ritsko, J. 1999. Personal communication with J. Ritsko, FPD Technologies, IBM, and J. Overly, University of Tennessee, Center for Clean Products and Clean Technologies. March.
- Socolof, M.L., J.G. Overly, L.E. Kincaid, D. Singh and K.M. Hart. 2000. "Preliminary Life-Cycle Assessment Results for the Design for the Environment Computer Display Project." Institute of Electrical and Electronics Engineers (IEEE) International Symposium on Electronics and the Environment, San Francisco, CA. May.
- Tekawa, M., S. Miyamoto and A. Inaba. 1997. "Life Cycle Assessment: An Approach to Environmental Friendly PCs." Institute of Electrical and Electronics Engineers (IEEE) International Symposium on Electronics and the Environment, San Francisco, CA. May.
- Tsuda, D. 1999. Personal communication with D. Tsuda, Apple Computer, and J. Overly, University of Tennessee, Center for Clean Products and Clean Technologies. July.

- Young, R. 1999. Personal communication with R.Young, Display Search, and J. Overly, University of Tennessee, Center for Clean Products and Clean Technologies. August.
- VP150. 1998. Viewsonic VP150 17" monitor specifications sheet.

Attachment A to Appendix H Supporting Tables this page intentionally left blank

°RTs	Company	Model	Size	VIS ^a			Energy Consumption (watts)						
	company		(inches)	(inches)		Full-on		Standby ^b Suspend Act. Off		Act. Off	Comments/Assumptions		
	Apple	Color Sync	17.0	16.1	<	125	<	60			<	5	
		Multiple Scan 720	17.0	16.0	<	120					<	5	
	Compaq	V75	17.0	16.0	<	115							
	(& Digital)	P75	17.0	16.0	<	115							
		71C	17.0	15.7	<	110	<	15	<	15	<	8	
		71P	17.0	16.0	<	120	<	15	<	15	<	8	
	EIZO	FlexScan TX-C7	17.0	15.8		140	<	12			<	5	
		FlexScan FX-C5	17.0	15.6		95	<	10			<	5	
	Hitachi	SuperScan Elite 641	17.0	15.9	<	135	<	15			<	8	
		SuperScan Pro 620	17.0	15.9	<	115	<	15			<	8	
	LG	Studioworks 74i	17.0	16.0		100		15				8	
		77M	17.0	15.9	<	130						5	
	MAG	XJ707	17.0		<	120	<	15	<	15	<	8	
		XJ717	17.0		<	120	<	10	<	10	<	5	
		XJ700T	17.0		<	120	<	15	<	15	<	5	
		DJ707 AV	17.0		<	120	<	15	<	15	<	8	
	Mitsubishi	Diamond 87TXM	17.0	16.0		120	<	100	<	15	<	8	Due to the significant difference in the Mitsubishi "Standby"
		Diamond Pro 700	17.0	16.0		110	<	95	<	15			power mode category values and those supplied by other
		Diamond Plus 72	17.0	16.0		105	<	90	<	15			"Standby" power mode calculation.
		Diamond Plus 70	17.0	16.0		95	<	80	<	15			
	NEC	Multisyne A700	17.0	15.6		85					<	8	NEC representative contacted through support phone number
		Multisync E700	17.0	15.6		95					<	8	stated that the energy saver mode power consumption is
		Multisync M700	17.0	15.6		120					<	8	similarities, assumed that this rating falls into the 'Active Off'
		Multisync P750	17.0	15.6		125					<	8	power mode consumption category.
	Panasonic	PanaSync S17	17.0	16.0		110	<	15	<	15	<	8	The company Web site stated "typical" or "nominal" for the
		PanaMedia PM17	17.0	16.0		130	<	20	<	20	<	8	associated power consumption values.
	Philips	107S	17.0	15.9		80					<	5	
		107MB	17.0	15.9		85					<	5	
		107B	17.0	16.0		85					<	5	
	Sony	CPD-200ES	17.0	16.0	<	120	<	15			<	8	
	-	CPD-200GS	17.0	16.0	<	120	<	15			<	8	
	Toshiba	TekBright 700P	17.0	15.8		100					1	1	Web site indicated 110 watts maximum, 100 watts nominal.
	Viewsonic	PT775	17.0	16.0		130							The company Web site stated "typical" or "nominal" for the
		EA771B	17.0	16.0		130							associated power consumption values.
		G773	17.0	16.0	<	110					1		
CRT Averages: 113.00				113.00		17.31		15.00		6.85			
Standard deviations: 15.35						11.61		2.13		1.49			
	CRT Standby, Suspend and Active Off average:									13.05			
Standard deviation						deviation:				5.08			

Table A-1. CRT & LCD Monitor Energy Consumption Values

			able AI.	U.	RIQI		D MIOIII	0 ľ	Energy	Ú	onsun	<u>iption values (continueu)</u>
LCDs	Company	Model	Size/VIS ^a			Ene	ergy Consun	npti	on (watts)			
			(inches)		Full-on		Standby ^b		Suspend		Act. Of	f Comments/Assumptions
-												
	Apple	Studio Display	15.1	<	35		10				8	
	Batron	FM-17TX11	15.0		35							
	Compaq	TFT500	15.0	<	50							
	Digital	51P	15.0		40		8				8	Received data through phone support (800.354.9000).
	EIZO	FlexScan L34	15.0		30	<	15			<	5	
	LG	500LC	15.1		40		5				5	
	Mitsubishi	LCD50	15.0		45		8				8	
	NEC	LCD1510	15.0		50		8					
	Samsung	500TFT	15.0	<	45					<	5	Received data via E-mail. The Samsung E-mail received stated that in full power on mode, the power consumption was a maximum of 45 watts and a nominal of 36 watts.
	Sharp	Super-V	15.0		36						2	
	Sony	CDP-L150	15.0	<	35	<	4			<	4	
	Viewsonic	VP150	15.0		35		2.6				2.6	Received data via E-mail. Full-on category value noted as "typical."
	LCD Averages: 3 Standard deviations:		39.67 6.50	9.67 7.58 5.29 .50 3.89 2.29					5.29 2.29			
	LCD Standby, Suspend and Active Stand			ve O dard	ff average: deviation:				6.44 3.09			

Table A1 CDT 9 LCD Maniton C 4. Val 1-. . . . **A**)

^a VIS = Viewable Image Size.

^b The 'Standby' energy consumption category includes listings noted as "Power Save Mode 1."

Notes: The energy consumption data shown in this table were taken from the Web sites of the retailers during 1998 unless otherwise noted. The energy consumption ratings for these monitors showed various information. Sometimes the less than (<) symbol preceeded some or all values, sometimes the addendum note 'maximum' was included and sometimes only the values themselves were reported.

MTBF Values for the:		s for the:	Source	Comments from Sources
CRT	CRT	LCD		
Monitor	Only	Monitor		
(thousand	hours)			
30-60	10-15		Goldwasser 1999	 "Most manufacturers will quote an MTBF of somewhere in the 30,000 to 60,000 hour range, EXCLUSIVE of the CRT. The typical CRT, without an extended-life cathode, is usually good for 10,000 to 15,000 hours before it reaches half its initial brightness." "CRT Life: The life of a monitor is determined by the life of the CRT. The CRT is by far the most expensive single part and it is usually not worth repairing a monitor in which the CRT requires replacement. The brightness half-life of a CRT is usually about 10-15k hours of on time independent of what is being displayed on the screen." "In a CRT monitor, the shortest-lived component BY FAR is the CRT itself, and it ages (more properly, the cathode is aging) as long as the heater is on and the tube is under bias (i.e., receiving voltage). Most monitors don't get around to turning the heater down or off until they enter the Display Power Management Signaling (DPMS) "suspend" or "off" modes. (And no, screen-savers do NOT help here - the tube is still on and the cathode is aging.) "In a CRT display, the CRT itself is usually the limiting factor in this (life), and in THAT specific case we usually speak of "mean time to half-bright" instead, since it's rare for a CRT to simply die once it's past its early operating life. Mean-time-to-half-bright is just what it says: how long, on average, can you operate the tube before the brightness drops to half its initial level for a given set of operating conditions. (Brightness is ALWAYS slow(ly) decreasing throughout the tube's life, due to the aging of the cathode and the phosphor.) For most tubes with standard cathodes, this will be in the neighborhood of 10,000-15,000 hours."
50-100			McConnaughe y 1999	Mr. McConnaughey stated that each of the subsystems of a monitor has different components that must meet different MTBF (Mean Time Before Failure) testing. Before testing, manufacturers typically calculate what the expected MTBF should be, and then test it to obtain the demonstrated MTBF. A rule of thumb is 50,000 hours calculated and over 100,000 hours demonstrated.
75			Philips 1998	"MTBF: >75,000 h (according to MIL-HDBK 217E) at 25 degrees Celsius (excl. CRT)"
50			Maginnovision 1998	"The average MTBF (Mean Time Before Failure) for MAG InnoVision monitors is 50,000 hours, excluding the CRT."
86			PlanetMac 1999	Mean Time Before Failure = 86,000 hours.
80	10-15	(50- backlights)	Douglas 1999	Phone conversation with David Douglas at Dell in Texas. David took plenty of time to discuss MTBF, and relayed that while Dell requires suppliers of CRT components (EXCLUDING THE CRT) to meet a MTBF specification of 80,000 hours, Dell performs testing (a type of 'demonstrated' MTBF - is a torture test) that typically yields at least twice the specification value in total time the equipment can operate. With that said, David then agreed that the CRT is the component that determines a CRT-based monitor's lifetime and that it is rare that a CRT lasts anywhere near that long, with most failing in the 10,000-15,000 hours/life range. David noted that CRT semiconductors are the next component that can fail. In LCDs, components containing silicon are most likely to fail first, with most manufacturers quoting backlights that will last 50,000 hours.
			Koch 1996	Didn't supply any other data other than that they assumed 10,000 hours as the lifespan of the LCD monitor.

Table A2. CRT and LCD Monitor MTBF Values & Manufactured Life Comments

М	TBF Value	es for the:	Source	Comments from Sources					
CRT Monitor	CRT Only	LCD Monitor							
(thousand	hours)								
		50: 15 for backlights	Tsuda 1999	Mr. Tsuda (Apple Computers) stated that the specs don't typically change for different size LCD monitors for specific components. MTBFs for flat panel displays are about 50,000 hours, except for the backlights which have MTBFs of about 15,000 hours. Most components can be fixed or replaced easily by trained technicians. Testing they perform is with maximum brightness, full white pattern; worst pattern for LCD is 1 pixel On/1 pixel Off.					
		40	Young 1999	Through a conversation with Ross Young, Ross spoke of a note a gentlemen had sent him wherein they assumed a useful life for an LCD of 84 months and a CRT of 36 months. Additionally, it was noted that an LCD panel was assumed to have a life of around 40,000 hours, and this could increase if DPMS screen savers were implemented.					
		(50- backlights)	VP150 1998	Light Source: long life, 50,000 hrs. (typ)					
		(10-40 - silicon driver chips)	Ritsko 1999	Liquid crystals and thin-film transistors (TFTs) don't typically wear out, yet the amorphous silicon transistors are less reliable than the single transistors. Also, the driver (silicon) chips could be an item that might show wear, however, the chips that go in FPDs are fairly typical, use low voltages and should run between 10,000 and 40,000 hours.					

Table A2. CRT and LCD Monitor MTBF Values & Manufactured Life Comments

References:

Douglas, D. 1999. Personal communication with David Douglas and Eric Korman of Dell. March.

- Goldwasser, S.M. 1999. "Notes on the Troubleshooting and Repair of Computer and Video Monitors." Web site available at: http://plop.phys.cwru.edu/repairfaq/REPAIR/F_monfaq.html.
- Koch, J. 1996. "Evaluating Environmental Performance: A Case Study in the Flat Panel Display Industry." Submitted in partial fulfillment of the requirements for the degree of Master of Science in the University of Michigan. April.

Maginnovision. 1998. Information found on a Web page from Maginnovision, available at: http://www.maginnovision.com.

McConnaughey, M. 1999. Personal communication with Mark McConnaughey, VP of Technology and Sourcing at ViewSonic. March.

Philips. 1998. c1995 Typhon 19"high resolution digital autoscan monitor (CRT) fact page. Web site available at:

http://www.fimi.philips.com/product/graphic/c1995.html.

Planetmac. 1999. Information found on a Web page from PlanetMac, available at: http://planetmac.com/Frames/catalog/monitors/gdm1962.html. Ritsko, J. 1999. Personal communication with John Ritsko, Manager, FPD Technologies at IBM. March.

Tsuda, D. 1999. Personal communication with Dani Tsuda with the Environmental Technologies and Strategies Department at Apple. July.

Young, R. 1999. Personal communication with Ross Young, founder and President of Display Search. August.

VP150. 1998. Viewsonic VP150 17" CRT monitor specifications sheet.

Author	Operating Circumsta		Power Consumption rate/Power Management	
Jung 1999	"The estimated useful life for a PC in a business envi	ronment is only two to three years, while	e home	1 8
9	computers users typically use their equipment for the	ree to five years."		
Matthews 1997	First research group to attempt to model the time eff during the Use life-cycle stage. In doing so, came up with a breakdo calculating the destinations and percent averages fro landfilled after 8.06 years in use and storage (and pos in use, storage and possible reuse.	nent ally, after aipment is 3.49 years		
	Initial lifetime of PCs: 5 years			
		% at end of 1st life recycled:	5%	
		% at end of 1st life stockpiled:	45%	
		% at end of 1st life landfilled:	5%	
	Lifetime of reused PC: 3 years	% reused recycled:	40%	
		% reused stockpiled:	50%	
		% reused landfilled:	10%	
	Lifetime of stockpiled PC: 3 years	% stored recycled:	75%	
		% stored landfilled:	25%	
	Calculating through the above numbers for 100 computers reveals the following breakdown:	(# out of 100)	(time)	
	Landfilled after 5 years	25		
	Landfilled after 8 years	124		
	Landfilled after 11 years	5.5	60.5	
	Landfilled totals	209.5		
	Average	8.06		
	Recycled after 5 years	25		
	Recycled after 8 years	52	416	
	Recycled after 11 years	17	187	
	Recycled totals	74	628	
	Avera	ge number of years to recycling of PC:	8.49	
NSC 1999	"The lifespan estimates used in this study were devel- than 30 major manufacturers and resellers. Major computer determine the lifespan of electronic equipment. Beca products were fabricated and many also have recycli to make an educated lifespan estimate. Resellers and to estimate the reusable life or 'second life' by produ- were used to develop estimates of the first life (the amount of time a product is useful to its original from manufacturers to disposal) for each electronic p	oped through interviews with more manufacturers were consulted to use manufacturers know when their ng facilities, these firms are qualified nonprofit organizations were asked ct and processor type. These inputs lowner) and the total lifespan (period product."	Total life	
	CRT computer	4	6-7	
	Notebook PC	2-3	4	

Author	Operating Circumstances/Lifetime Data		Power Consumption rate/Power Management	
Chan 1997	This class report from a University of Toronto group o information. The data presented comprise responses f research staff and 50 residents). Class covered 1996/19	f 4 people contains several worthwhile ₁ rom 180 people (130 administrative, tea 997; data were gathered during the class	pieces of ching and s.	
	Hours of computer use by staff	< 4 hrs/dy	7%	
		5-8 hrs/dy	52%	
		9 or > hrs/dy	41%	
	Percentage of computers with energy-saving features	have features	52%	
	installed or activated	no knowledge of features	35%	
		don't have features installed or	13%	
	Respondents who update their knowledge of	do not	75%	
	computer energy-saving features	do	19%	
		no response	6%	
	Idling time of office and residential computers that	less than 2 hrs	66%	
	are turned on	3-5 hrs.	22%	
		6 hrs or more	12%	
	Respondents who turn off their computer when they	never do	70%	
	are away for a period (period is 45 min. or longer)	sometimes do	21%	
		always do	7%	
		no control	25%	
	Staff who shut down their computers at the end of	always	70%	
	the day	sometimes	9%	
		never	19%	
		no control	2%	
	Percent of office computers left on during weekends	always	22%	
		sometimes	8%	
		never	67%	
		no control/no response	3%	
EIA 1997	The EIA's results from the Residential Energy Cons	umption Survey (RECS) provides some	good data on	
	Hours PC turned on each week	less than 2 hrs	8.2	
		2 to 15 hrs	17.4	
		16 to 40 hrs	6.7	
		On all the time	3.3	
	How PC is used	15 hrs a week or less	26.5]
		16 hrs a week or more	10.0]
		Personal use only	4.8	

Author	Operating Circumstances/Lifetime Data	Power Consumption rate/Power Management							
		Business use only	2.1						
		Used for both	3.1						
	Additionally, RECS calculated that lion computers we from the RECS included "6% of the households that u	her data mute."							
CIA 1998	Estimate that 117 million computers were in use in the								
EPA 1999	Have the EnergyStar compliance monitor specification	ns.							
		First low-power mode	Second						
	Low-power state:	< = 15 watts	<=8W						
	Default times:	15-30 minutes	< 70 min.						
Koch 1997	Assumed 10,000 hours/lifetime for the LCDs in the stu	ıdy.							
Goldberg 1998	Article reported that Walt Rosenberg, Compaq's dire machines have a useful life of two-to-three years.	ctor of environmental affairs, stated tha	t today's						
Miseli 1999	Did not separate office from home user. Assumed units operate 50% of the time (annually) around the clock. Stated that "True life of a CRT or LCD is defined for the case when it runs continually at full intensity," adding that true life for CRT is about 1.25 years and for LCD is about 2.9 years. Doubled each of those true life values for his calculations.								
Tekawa 1997	Assumed personal users time frame of 2 hr/dy, 365 dy/yr for 5 years, and office users time frame of 8 hr/dy, 247 dy/yr for 7 years. Assumed a ratio of personal to office user of 4.6. Don't state the actual numbers they used, but do say they took the mean of the minimum and maximum power consumption ratings.								
Atlantic 1998	They estimated that a PC's lifetime is 3 years. Then they stated that they were modeling only the first lifetime of a PC; they acknowledged other lifetimes but decided not to attempt to model them. They also estimated that the PC is turned on 8 hrs per day, 230 days per yr, altogether running for 5,520 hrs during its lifetime. They also a case PC has no energy savings facilities."								
Philips 1998	MTBF of 75,000 hrs for a 19" C1995 Typhoon high re	esolution CRT monitor excluding the CF	RT.	Power consumption: 120W typ. (140 W max)					
Nordman 1996	In this document, Lawrence Berkley National Labora performed determining the state of power consumption monitors	tory (LBNL) details results from several n and power management in certain co	l audits they mputers and	The LBNL document provided results from a audit of 70 monitors and their setup and use of energy saving power modes. Their primary conclusions were that only approximately one-third of all monitors were "accomplishing power management." The following is a breakdown of some of what they found:					
	Standard % of time in each mode by day type, by operating pattern	Full-on	Low	- 34 apparently meet Energy Star requirements					
	Workday	17%	35%	- 30 were 'universal,' (able to initiate power mgmt two ways)					
	Weekend day	0%	20%	- 30 were left on at time of audit (12 in suspend mode)					
	Absence day	0%	20%						
	Weekdays average	13%	45%						
	All days average	10%	35%						
				This document also contained data on the actual power consumed by 3-17" monitors over a 4-6 week period, broken down by power consumption mode, and the results are shown below:					
				Monitor #1: Full-on = 91 watts; Low = 7 watts					
				Monitor #2: Full-on = 84 watts; Low = 3 watts					
				Monitor #3: Full-on= 85 watts; Low = 4 watts					

Author	Operating Circumstances/Lifetime Data	Power Consum	ption rate/Power	Management	
CCPCT 1998		The University of available CRT and (mostly via the WV consumption brea consuming state.	The University of Tennessee CCPCT reviewed the available CRT and LCD energy consumption information (mostly via the WWW) and produced the energy consumption breakdown shown at left by energy consuming state. The units are all watts.		
		CR	Full-on:	113.29	
			Standby:	17.18	
			Active off:	6.85	
		LCI	Full-on:	40.00	
			Standby:	7.58	
			Active-off:	5.70	

References (not listed in the main References list):

Atlantic Consulting. 1998. "LCA study, EU Ecolables for Personal Computers." January.

- CCPCT (Center for Clean Products and Clean Technologies). 1999. Calculated averages of energy use data found from multiple computer manufacturers's Web sites.
- Chan, A., et al. 1997. "The role of Computers in Energy Consumption on Campus." A class research paper from the University of Toronto. Obtained copy from the Internet at: http://server.esc.cquest.utoronto.ca/env/env421h/energy/computers.html.

EPA (Environmental Protection Agency). 1999. Energy Star-labeled office equipment with computer specifications. Web site available at: http://www.epa.gov/appdstar/esoe/pcsum.html. Goldberg, L. 1998. "The Advent of "Green" Computer Design." *Computer Magazine*. September.

- Jung, L.B. 1999. "The Conundrum of Computer Recycling." Resource Recycling. May.
- Koch, J. 1996. "Evaluating Environmental Performance: A Case Study in the Flat Panel Display Industry." Submitted in partial fulfillment of the requirements for the degree of Master of Science in the University of Michigan. April.

Matthews, H.S., et al. 1997. "Disposition and End-of-Life Options for Personal Computers." Report available online from: June. http://www.ce.cmu.edu/GreenDesign/comprec/index.html.

Miseli, J. 1999. "Hidden Cost Savings of Flat Panel Displays." Display Manufacturing Technology Conference Digest of Technical Papers, San Jose, CA. February.

Philips. 1998. c1995 Typhon 19" high resolution digital autoscan monitor (CRT) fact page. Web site available at: http://www/fimi/philips.com/product/graphic/c1995.html.

Tekawa, M., S. Miyamoto and A. Inaba. 1997. "Life Cycle Assessment, An Approach to Environmentally Friendly PCs." Institute of Electrical

and Electronics Engineers (IEEE) International Symposium on Electronics and the Environment. San Francisco, CA. May.

APPENDIX I

TECHNICAL MEMORANDUM: End-of-Life Approach for the DfE Computer Display Project

this page intentionally left blank

APPENDIX I

TECHNICAL MEMORANDUM: End-of-Life Approach for the DfE Computer Display Project

I. INTRODUCTION

The Computer Display Project, sponsored by the U.S. Environmental Protection Agency (EPA) as part of its Design for the Environment (DfE) program, is investigating the life-cycle impacts of cathode ray tube (CRT) displays and liquid crystal displays (LCDs) for use in desktop computers.

A meaningful comparison of the two technologies from the environmental perspective can be made only if the life-cycle impacts associated with various stages, namely, raw material extraction, materials processing, manufacture, use, and end-of-life (EOL), are evaluated. The functional units being compared are the 17-inch CRT display monitor and the 15-inch active matrix liquid crystal display (AMLCD) monitor. The two are considered to be functionally equivalent with respect to the viewing area available to the user. EOL issues are of growing interest to manufacturers nowadays due to Extended Producer Responsibility (EPR) concerns (Fishbein 1998) and the consequent higher expectations from manufacturers for influencing the ultimate fate of their products.

The purpose of this memorandum is to describe the approach for evaluating the EOL life-cycle stages of CRTs and AMLCDs for the Computer Display Project. This approach includes:

- (1) developing scenarios to represent reasonable EOL alternatives; and
- (2) collecting life-cycle inventory (LCI) data for the EOL alternatives.

1.1 Background

Estimates from 1998 revealed that more than 20 million personal computer central processing units (CPU) became obsolete in that year (NSC 1999). Earlier estimates indicated that approximately 10 million television sets and 12 million computer monitors reach the end of their useful lives each year (MCC 1996). Assuming that one monitor became obsolete for every CPU in 1998, and since LCDs have not been in existence long enough to have attained "end-of-life" (EOL) status in sufficiently large numbers, it is expected that approximately 20 million CRTs are retired annually. There is not much information available on the disposition options for LCDs.

The major existing EOL environmental concern associated with CRTs is disposal of leaded glass. LCDs, on the other hand, do not contain any leaded glass and are much lighter in weight, but contain other materials of concern, such as mercury used in the backlights.

1.1.1 CRT EOL Issues

According to practices followed by leading CRT recyclers such as Envirocycle (Envirocycle 1999), the material of greatest value recovered from CRTs is leaded glass (which is also the major component by weight), followed by small quantities of metals. Also, whenever the incoming EOL product is a complete computer monitor, some plastics and metals can be recovered from its outer casing and other parts.

The closed loop recycling of leaded glass involves recovering and processing the material for use as cullet in the manufacture of new CRTs. CRT manufacturers will use the cullet if it meets quality standards and is of the same chemistry and type as required by them in their manufacturing operations. Thus, effective recycling of post-consumer CRT glass requires very careful sorting and separation into various types, followed by decontamination (removal of coatings). Resmelting (for lead recovery) and downcycling (into other glass applications) are some of the other "open loop recycling" alternatives. In 1997 and 1998, CRT computer monitor recycling was done for 1.3 million units (46 million pounds) and 1.5 million units (51 million pounds), respectively (NSC, 1999).

1.1.2 Regulations Regarding CRT Disposal

Color CRTs may fail the EPA Toxicity Characteristic Leachate Procedure (TCLP) test, and therefore may be classified as hazardous waste under current EPA regulations. Some experts believe that this classification poses barriers to the effective recycling of CRTs, on account of special permits and transportation requirements for handling hazardous waste (EPA-CSI 1999). However, EPA has implemented a glass-to-glass recycling exception.

In order to landfill CRTs in accordance with EPA regulations, they must be dismantled, and the glass crushed and stabilized by micro-encapsulation in cement. However, this method has some drawbacks. Crushing increases surface area and, consequently, the potential to leach lead. Though cement encapsulation is the required method, it has been found that cement disintegrates faster than glass (MCC 1994).

To encourage recycling, some states have developed new initiatives that will ease some of the regulatory barriers. Massachusetts, for example, has proposed to specifically exempt "intact" CRTs from being classified as hazardous waste and simultaneously banned them from disposal in municipal landfills and combustion facilities (MDEP 1999). These measures could promote the recycling of CRTs by making the process of handling and transportation much easier, and the paperwork less cumbersome.

1.1.3 LCD EOL Issues

Currently, no infrastructure or established process exists for recycling LCDs specifically. Of the small numbers of LCDs that have reached the EOL stage (predominantly as notebook computers), a much smaller number is likely to have reached recycling facilities. No specific details are available on the materials recovered from them as they are expected to have been processed along with other electronic products, with some valuable and/or potentially recyclable materials removed. The following components and materials of potential reuse or recycling value found in LCDs have been identified by MCC (MCC 1994):

- Thin film transistors (TFTs).
- Color filters.
- Glass.

The toxicity potential of heavy metals is of concern in the EOL stage. The heavy metals found in the LCD monitors are identified in the main body of this LCA report. This study will consider the presence of heavy metals or other materials of potential concern in the wastes and emissions generated.

1.2 EOL Disposition Options

In the past, landfilling has been the prevalent method for the disposition of post-use computer monitors (i.e., those re-used after being resold or donated). However, with increasing awareness of potentially harmful life-cycle environmental impacts, dwindling natural resources, government regulations against disposal of toxic substances in landfills, and the consequent development of markets for recycled components and materials, more options are now available for the disposition of post-use computer monitors. They are briefly described below.

1.2.1 Reuse

Reuse, often as a result of reselling, involves continued use of the monitor for the purpose for which it was built, and is considered to occur within its originally intended useful life. Reuse does not usually entail major repairs or modifications, and is a preferred EOL option because the original materials contained in it are put to use for an extended period of time, thus conserving valuable natural resources (energy and raw materials) needed to manufacture new monitors or to dispose of discarded ones. However, reuse could result in reduced energy efficiency during the use stage as monitor manufacturers continually strive to improve the energy efficiency of their products.

1.2.2 Remanufacturing

Remanufacturing is a viable option for monitors that are no longer functional but could be refurbished (upgraded or restored to working conditions) at a cost lower than that of manufacturing a new monitor, to be sold again in domestic or foreign markets.¹ Here again, energy and raw materials are conserved, though some new parts/components may be required. Another important benefit of remanufacturing is solid waste reduction, achieved by diverting the monitor materials away from the landfill. Remanufacturing processes span a wide range of activities, from as little as replacing button tops to as extensive as testing and replacing PCBs or transformers.

¹ In addition to cost, the arrival of new technology is another factor that inhibits remanufacturing. In such cases, remanufacturers seek to find markets where products based on old technology are still in demand.

1.2.3 Recycling

Recycling involves recovering the individual materials from EOL monitors, to be used in the production of new monitors (closed-loop recycling) or in other products (open-loop recycling). Identification, sorting, cleaning, and further processing (e.g., smelting) are often required before the recovered materials can be used again. Though materials recycling involves several processing steps, it results in the conservation of energy and raw materials, and diversion of materials that would otherwise have been landfilled, through the creation of new, desirable products that are in-line with current market demand.

1.2.4 Waste-to-Energy (WTE) Incineration

A portion of municipal solid waste (MSW) is routinely sent to incinerators or municipal waste-to-energy (WTE) facilities for energy recovery. The quantity of ash (bottom ash and fly ash) left over is a small fraction, around 25% (EPA 1998) of the original waste input, and can be disposed of either as non-hazardous or hazardous waste, depending on whether it passes the TCLP test or not. The obvious benefits are reduced solid waste and the energy produced, which is often counted as a credit in life-cycle energy calculations.²

1.2.5 Landfilling

Landfilling solid waste in Subtitle C (for hazardous) or D (for non-hazardous) landfills is the least preferred option, since all the other options have some expected environmental benefits. The disposal of waste in Subtitle C landfills is usually the most undesirable, as it often involves treatment to immobilize the hazardous materials before they can be landfilled, thus increasing the quantity and cost of disposal. Also, hazardous waste sites have the potential to turn into high liability ("Superfund") sites. Some states have regulatory activities that might not accept monitors in Subtitle D landfills.

2. METHODOLOGY

This section outlines key assumptions, defines conceptual models proposed for determining the flow of materials through the EOL processes, and highlights some important issues pertaining to CRTs and LCDs.

The major steps are listed below:

- Assumptions about the distribution of EOL options were made.
- Data were collected for various disposition options using existing inventory reports and inventory questionnaires sent to recyclers.
- Data were normalized to the functional unit and included as the EOL inventories.

 $^{^2}$ The energy used in different life-cycle stages is summed up to arrive at the total energy used in the life cycle of the product. In case of WTE incineration, where energy is recovered instead of being used up, it is treated as a negative value and subtracted from the total.

2.1 EOL Conceptual Models

A monitor is assumed to have reached EOL status when:

- It has served its useful life and/or is no longer functional.
- Technological obsolescence renders it unusable.

The EOL options for CRT and LCD monitors are graphically depicted in Figure 1. Estimates of the percent distribution of monitors going to each EOL option are presented below. As the functional unit in this study is one monitor over its lifetime, the percentages are used as probabilities for the EOL disposition of a particular monitor.



Figure 1. Conceptual Model Showing End-of-Life Disposition Options for CRT and LCD Monitors

2.2.1 CRT

The National Safety Council (NSC 1999) reported that 11% of all personal computer CPUs are recycled. Assuming one monitor is recycled with every CPU, and assuming these represent CRTs, we assume 11% of CRTs go to recycling. The NSC report also stated that 3% of personal computers are "refurbished and resold or donated." We thus used 3% as an estimate of the CRT monitors that are remanufactured, although we recognize this might be an overestimate, as we do not know if those resold and donated are also remanufactured. Given that this is a small percentage, this error is not expected to have a large effect. Further data are lacking on the percent of monitors being incinerated or going to landfills. To estimate the percent incinerated, we used the percent of all municipal solid waste in the United States being incinerated, which is estimated at 15% (EPA 1998). Summing the percents for recycling, remanufacturing, and incineration equals 29%. This leaves 71% that is assumed to be landfilled. In the life-cycle analysis in this study, only one landfilling process is modeled, which is assumed to represent both hazardous waste and solid waste landfilling. The landfilling process is derived from Ecobalance data and is a combination of four major materials in a CRT (glass, steel, plastic, and aluminum), based on the proportion of each of those materials in the CRT. The inventories for each material are of generic materials (not necessarily the precise materials in the CRT). For example, the glass is generic glass, and not leaded glass, and the plastics are generic "plastic" and may not represent the exact plastics in the CRT.

Although the percentage of monitors that are landfilled are not separated into hazardous and non-hazardous waste landfilling processes, we have still attempted to estimate the proportion of CRTs that go to each landfill. Due to a lack of data, we assumed as a best estimate that the percent of monitors that are in households are equivalent to the percent of landfilled monitors that would be disposed of in a solid waste (Subtitle D) landfill and the percent of monitors that are in businesses would be disposed of in a hazardous waste (subtitle C) landfill. As presented in the Use Stage discussion in the main body of this report (Section 2.4.1.2), 35% of monitors are in households and 65% are in office and other environments. Therefore, of the 71% of monitors assumed to going to landfills, 25% are assumed to be sent to solid waste landfills and 46% to hazardous waste landfills. To summarize, the EOL dispositions assumed for the CRT are as follows:

•	Incineration:	15%
•	Recycling:	11%
•	Remanufacturing:	3%
•	Hazardous waste landfill:	46%

• Solid waste landfill: 25%

2.2.1 LCD

Data were even more lacking for the EOL dispositions of LCDs. The same 15% of municipal solid waste incinerated in the United States was assumed for LCD incineration as it was for the CRT. An individual in the monitor recycling business estimated that no more than 5% of LCDs are sent to hazardous waste landfills and that essentially none are currently being recycled (Vorhees 2000). Given this limited data, the remaining 80% needed to be split between

solid waste landfilling and remanufacturing. Given no other data, we assumed half of the remaining 80% goes to solid waste landfills and half to remanufacturing. Assuming that 40% are remanufactured is likely an overestimate; however, no supporting data were available to modify this estimate. Therefore, in the baseline analysis of this study, the following percentages have been used:

- Incineration: 15%
- Recycling: 15%
- Remanufacturing: 15%
- Hazardous waste landfill: 5%
- Solid waste landfill: 50%

Sensitivity analyses have been conducted to determine the effects of these assumptions on the results and are discussed in Chapters 2 (LCI) and 3 (LCIA) of the main report.

2.2 Assumptions

In developing the EOL model for CRT and LCD monitors, it was necessary to make several assumptions. In addition to the percent distributions presented above, the following assumptions apply to the EOL data:

- Reuse and resale are not included in the EOL scenarios modeled as these events are considered to occur within the originally intended useful life of the monitor.
- The monitors currently in storage are not considered to have reached EOL yet and have, therefore, been excluded from the EOL model. Moreover, they are assumed not to have environmental impacts while in storage, and maintenance of storage space is assumed to the beyond the scope of this study.
- Only waste-to-energy incineration is modeled because there is very limited straight incineration (without energy recovery) being done in the U.S. at present. In fact, the 1996 total U.S. WTE design capacity was 100,355 tons per day, with 110 WTE facilities in operation. In contrast, the capacity for incineration without energy recovery was only 2,451 tons per day, with a total of 19 facilities in operation. In general, WTE has become the prevalent method for MSW combustion since the 1980s (EPA 1998).

REFERENCES

- CIA (Computer Industry Almanac). 1999. Web site available: http://www.c-i-a.com/19961225.htm . April 30.
- Denison, R. 1996. "Environmental Life-Cycle Comparisons of Recycling, Landfilling, and Incineration." R.A. Denison, Environmental Defense Fund.
- EIA (Energy Information Administration). 1999. Web site available: http://www.eia.doe.gov/emeu/recs/recs97/appusage.html#pcusage. April 30
- Envirocycle. 1999. Web site available: http://www.enviroinc.com/. March 8.
- EPA (Environmental Protection Agency). 1998. "Characterization of Municipal Solid Waste in the United States: 1997 update." U.S. EPA Office of Solid Waste, Report No. EPA530-R-98-007. Prepared by Franklin Associates. May.
- EPA. 1998. "Characterization of Municipal Solid Waste in the United Sates: 1997 Update." Report No. EPA 530-R-98-007.
- EPA. 1999. EPA website 4/26/99. http://www.epa.gov/ooaujeag/taskforce/issppr1.htm
- EPA-CSI. 1999. U.S. EPA's CSI web pages 4/26/99. http://www.epa.gov/csi/computer/accomp6.htm
- Fishbein, B.K. 1998. "EPR: What Does It Mean? Where Is It Headed?" *Pollution Prevention Review.* 8 (4): 43-55. John Wiley & Sons, Inc. October.
- Laboratory Network. 1999. Web site available: http://news.laboratorynetwork.com/regulatory-news/19981211-6948.html. April 26.
- MCC (Microelectronics and Computer Technology Corporation). 1994. *Electronics Industry Environmental Roadmap.* Chapter 7.
- MCC. 1996. "1996 Electronics Industry Environmental Roadmap." June.
- MDEP (Massachusetts Department of Environmental Protection). 1999. Web site available at: http://www.state.ma.us/dep/bwp/dswm/files/crtq&a.htm. March 12.
- NSC (National Safety Council). 1999. Electronic Product Recovery and Recycling Baseline Report: Recycling of Selected Electronic Products in the United States. Washington, DC, National Safety Council. May.
- RTI (Research Triangle Institute). 1999. Personal Communication with K. Weitz. "EOL Management Options for MSW." February.

- Scott, M.H., F.C. McMichael, C.T. Hendrickson, and D.J. Hart. 1997. "Disposition and End-of-Life Options for Personal Computers." Green Design Initiative Technical Report #97-10, Carnegie Mellon University.
- Tufts University. 1996. "Life Cycle Assessment of the Disposal of Household Electronics." A Capstone Thesis Project for M.S. in Civil and Environmental Engineering, submitted by D. McKenna, K. Powers, and S. Reagan, p. 5-6. August.

this page intentionally left blank

ATTACHMENT A TO APPENDIX EOL: EOL QUESTIONNAIRE

this page intentionally left blank

UF

DESIGN FOR THE ENVIRONMENT COMPUTER DISPLAY PROJECT Life-Cycle Inventory (LCI) Data Collection Questionnaire for the End-of-Life Stage



Introduction

The Design for the Environment (DfE) Program in the U.S. Environmental Protection Agency's (EPA) Office of Pollution Prevention and Toxics has begun a voluntary, cooperative project with the electronics industry to assess the life-cycle environmental impacts of cathode ray tube (CRT) and liquid crystal display (LCD) desktop monitors. The DfE Program conducts comparative analyses of alternative products or processes to provide businesses with data to make environmentally informed choices about product or process improvements. The DfE Program has no regulatory or enforcement agenda and was established to act as a partner with industry to promote pollution prevention. This environmental life-cycle assessment will address human and ecological risk, energy and natural resource use, performance, and cost of various display technologies. The University of Tennessee (UT) Center for Clean Products and Clean Technologies is conducting the life-cycle inventory (LCI), which is the data collection phase of a life-cycle assessment, with technical assistance from Microelectronics and Computer Technology Corporation, the Electronics Industry Alliance, and other partners.

Boundaries

A *life-cycle* assessment considers impacts from materials acquisition, material manufacturing, product manufacturing, use, and final disposition of a product. The LCI data are intended to be used to evaluate relative environmental impacts over the entire life-cycle of a product, including transport between life-cycle stages. In this project, the product is either a color CRT or LCD monitor. Therefore, data associated with the materials and processes used directly in the manufacturing, use, and disposition of the product are relevant to the LCI and requested in this questionnaire. Please include only materials or energy *directly* used in the disassembly, remanufacturing, recycling, or disposal of the monitor or its components (e.g., *do not include* general building heating and air conditioning).

Product focus

This project focuses on 17" CRT and 15" LCD desktop monitors. We will appreciate your providing data specifically on these sizes, to the extent possible.

INPUTS: OUTPUTS:		OUTPUTS:	
Primary materials		$ \longrightarrow$	Products
Ancillary materials	Process	 —→	Air emissions
Utilities		$ \longrightarrow$	Releases to land
		$ \longrightarrow$	Water effluent

Fig. 1. End-of-Life process inventory conceptual template

Inventory data

We are asking you for data on CRT and LCD desktop monitors that you either remanufacture, or disassemble and recover, reuse, or recycle components and

materials from. The inputs and outputs data (Fig. 1) that you provide will be aggregated in the LCI to quantify the overall inputs and outputs of a CRT and LCD. Additionally, transportation information is requested in the inventory.

EOL Data Collection Form - p. i

Data sources

Much of the requested information can be drawn from existing sources, including, but not limited to the following:

- 1. Purchase and production records
- 2. Bills and invoices
- 3. Material Safety Data Sheets (MSDS)
- 4. Toxic Release Inventory (TRI) forms

6. Local, state, and federal reporting forms (e.g., hazardous waste manifests)7. Local, state, and federal permits8. Monthly utility billing records

5. Audit and analysis results (e.g., wastewater discharge analyses)

How the data will be used

UT will collect inventory data and tally the inputs and outputs for the different monitors. Information gathered by these questionnaires will be used to develop environmental profiles based on inputs and outputs for each stage in the life cycle of displays. The profiles will be used to evaluate environmental impacts from each product. Cost data will also be collected and presented along with environmental results. The environmental profiles can be used to encourage product design changes for product improvement. UT will aggregate data and ensure that data associated with particular companies remain anonymous to the EPA. UT can enter into confidentiality agreements where proprietary data are concerned. Please understand that accurate and representative information from you is critical for the success of this project.

Results of project

The results are intended to provide industry with an analysis of the life-cycle environmental impacts, cost, and performance of CRT and LCD computer monitors. Results will help identify areas for product and process improvement as related to risk and environmental impact (e.g., identifying material use inefficiencies) and will identify impacts from various life-cycle stages of the product systems. Use of the results will also help meet growing global demands of extended product responsibility.

Benefits of involvement

Your input will allow for your interests to be considered in the project development and data collection. By supplying data, the results will partially reflect your operations and, therefore, the results will be directly relevant to your interests. The project will allow you to directly apply results to your own processes and identify areas for improvement. You will also be recognized as working voluntarily and cooperatively with the U.S. EPA.

Deadline

Please complete this form and return it to us at the address below by September 30, 1999. If this is not possible, please contact Maria Leet Socolof at 423-974-9526 or at the addresses below to discuss alternative dates.

Your cooperation and assistance are greatly appreciated.

For any questions, please contact Maria Leet Socolof at 423-974-9526, <socolofml@utk.edu> or Rajive Dhingra at 423-974-8752, <rdhingra@utk.edu> at the University of Tennessee, 311 Conference Center Bldg., Knoxville, TN 37996-4134. For more project details, see the Project Fact Sheet, DfE Website <http://www.epa.gov/opptintr/dfe/compdisp/compdisp.html>, or the Draft Final Goal Definition and Scoping Document.

EOL Data Collection Form - p. ii

INSTRUCTIONS

- 1. Please be sure to read the introductory text on each page before filling out the questionnaire.
- 2. The data you supply in the tables should represent inputs and outputs associated only with the "product of interest" (i.e., materials, components or subassemblies that are either part of, or that are itself, the desktop monitor as defined on p. i under Product focus). If quantities provided are not specific to the "product of interest," please explain how they differ in the comments section at the bottom of the appropriate table.
- 3. Where supporting information is available as independent documents, reports or calculations, please provide them as attachments with reference to the associated page(s) or table(s) in this questionnaire.
- 4. If you have more than one product of interest to this project, please duplicate this questionnaire and fill out one questionnaire for each product.
- 5. If there is not adequate room on a page to supply your data (including comments), please copy the appropriate page and attach it to this packet.

6. The ensuing pages refer to the four indices shown below to detail specific information about the data. Additional information is provided below as required. Data Quality Indicators Index: These indicators will be used to assess the level of data quality in this questionnaire. Please report a DQI for the numerical value requested in each table on the following pages. The first category, Measured, pertains to a value that is a directly measured quantity. The second category, Calculated, refers to a value that required one or more calculations to obtain. The third category, Estimated, refers to a value that required a knowledgable employee's professional judgement to estimate. Lastly, the fourth category, Assumed, should be used only when a number had to be guessed.

Hazardous and Nonhazardous Waste Management Methods Index: These methods are applicable to both hazardous and nonhazardous wastes (Tables 8a and 8b). Please give the appropriate abbreviation in the Management Method column on p. 8 where requested. Depending on whether the management method is on or offsite. please indicate by specifying "on" or "off" in the appropriate column on p. 8.

For Tables 2, 3a, 3b, 4, 5, 8a, and 8b:

- **Transportation Modes Index** A - Large truck (18-wheeler), diesel
- B Small truck, diesel
- C Small truck, gasoline
- D Rail, diesel
- E - Barge, diesel
- F - Ocean freighter, diesel
- G - Other (please specify in comments section)

For Table 7b:

Wastewater Treatment/Disposal Methods Index

- A Direct discharge to surface water
- B Discharge to offsite wastewater treatment facility
- C Underground injection
- D Surface impoundment (e.g., settling pond)
- Direct discharge to land E
- F - Other (please specify in comments section)

For Tables 3a, 3b, 4, 5, 6, 7a, 8a and 8b:

Data Quality Indicators Index		
M	- Measured	
С	- Calculated	
Е	- Estimated	
Α	- Assumed	

For Tables 8a and 8b

CHATAPADADE ANA WANNATAFAAME WAREN WANDAAMANI WAIDAKE IADAT

- RU - Reused
- Recycled R
- Landfilled L
- Iv - Incinerated - volume reduction
- Incinerated energy conversion Ie
- S - Solidified/stabilized
- D - Deep well injected
- 0 - Other (please specify in comments section)

IF YOU HAVE QUESTIONS, PLEASE CONTACT EITHER:

OR

Maria L. Socolof (Project Manager): Phone: 423-974-9526 Email: socolofml@utk.edu

Rajive Dhingra (Project Engineer): Phone: 423-974-8752

Email: rdhingra@utk.edu

EOL Data Collection Form - p. iii
APPENDIX I

able 1.	Facility Information		Contact Information	
1. Company name:		5a. Prepared by:		Date:
2. Facility name:		5b. Title:	·····	
Facility address (location):		5c. Phone number:		Ext.:
	· · · · · · · · · · · · · · · · · · ·	5d. Fax number:		
		5e. Email address:		
	· · · · · · · · · · · · · · · · · · ·	·		
· · · · · · · · · · · · · · · · · · ·				· · ·
I. Products handled onsite:				
			t i i i i i i i i i i i i i i i i i i i	

EOL Data Collection Form - p. 1

• •

2. PRODUCT OF INTEREST INFORMATION

Product of interest:	E.g., End-of-Life CRT/LC	D desktop monitor		
e. No. of monitors processed annually (e.g., units, kg, lbs):				
. Facility's percent global market				
. Product of interest				n na katalan ka
um weight.	чи, <u>и ладина во о</u> бласти и и ладина и у ладина и од на области и области и области и области и области и области			
Brief description of the main operations/subproc	cesses			
requires to process the product of interest.				· · · · · · · · · · · · · · · · · · ·
•		·	- ·	
				· · ·
. Product transport information. In this table, ple modes' abbreviations. Percent capacity represer	ase list the top five locations (b	by quantity) from where you receive the products	ct of interest. See Transportation	Modes Index on p
. Product transport information. In this table, plea modes' abbreviations. Percent capacity represer Location (City, State)	ase list the top five locations (to the transport of the	by quantity) from where you receive the product t vehicle's total load was carrying the products Number of trips annually	ct of interest. See Transportation of interest. Percent capacity	Modes Index on p
Product transport information. In this table, ple modes' abbreviations. Percent capacity represer Location (City, State)	ase list the top five locations (t its what percent of the transpor Mode	by quantity) from where you receive the product rt vehicle's total load was carrying the products Number of trips annually	ct of interest. See Transportation of interest. Percent capacity	Modes Index on p
. Product transport information. In this table, ple modes' abbreviations. Percent capacity represer Location (City, State)	ase list the top five locations (t its what percent of the transpor Mode	by quantity) from where you receive the product t vehicle's total load was carrying the products Number of trips annually	ct of interest. See Transportation of interest. Percent capacity	Modes Index on p
. Product transport information. In this table, plea modes' abbreviations. Percent capacity represer Location (City, State)	ase list the top five locations (b its what percent of the transpor Mode	by quantity) from where you receive the product t vehicle's total load was carrying the products Number of trips annually	ct of interest. See Transportation of interest. Percent capacity	Modes Index on p
Product transport information. In this table, plea modes' abbreviations. Percent capacity represer Location (City, State)	ase list the top five locations (top five locations) (top five locations	by quantity) from where you receive the product t vehicle's total load was carrying the products Number of trips annually	ct of interest. See Transportation of interest. Percent capacity	Modes Index on p
Product transport information. In this table, ple modes' abbreviations. Percent capacity represer Location (City, State)	ase list the top five locations (t its what percent of the transpor Mode	by quantity) from where you receive the product t vehicle's total load was carrying the products Number of trips annually	ct of interest. See Transportation of interest. Percent capacity	Modes Index on p
. Product transport information. In this table, ple modes' abbreviations. Percent capacity represer Location (City, State)	ase list the top five locations (b nts what percent of the transpor Mode	by quantity) from where you receive the product t vehicle's total load was carrying the products Number of trips annually	ct of interest. See Transportation of interest. Percent capacity	Modes Index on p
Product transport information. In this table, plea modes' abbreviations. Percent capacity represer Location (City, State)	ase list the top five locations (to the transportion of the transport of the transportion of the transport of the t	by quantity) from where you receive the product t vehicle's total load was carrying the products Number of trips annually	ct of interest. See Transportation of interest. Percent capacity	Modes Index on p
Product transport information. In this table, ple modes' abbreviations. Percent capacity represer Location (City, State)	ase list the top five locations (top five locations) (top five locations	by quantity) from where you receive the product t vehicle's total load was carrying the products Number of trips annually	ct of interest. See Transportation of interest. Percent capacity	Modes Index on p
Product transport information. In this table, ple modes' abbreviations. Percent capacity represer Location (City, State)	ase list the top five locations (to the transportion of the transport	by quantity) from where you receive the product to vehicle's total load was carrying the products Number of trips annually	ct of interest. See Transportation of interest. Percent capacity	Modes Index on p
. Product transport information. In this table, ple modes' abbreviations. Percent capacity represer Location (City, State)	ase list the top five locations (b its what percent of the transport Mode	by quantity) from where you receive the product t vehicle's total load was carrying the products Number of trips annually	ct of interest. See Transportation of interest. Percent capacity	Modes Index on p

APPENDIX I

3. PRIMARY & ANCILLARY INPUTS

- 1. Primary & Ancillary Materials: Primary materials are defined as those materials that become part of a product output. Ancillary materials are those material inputs that assist in a process, yet do not become part of the final product. Please include the trade name and the generic name of each material where applicable.
- 2. CAS # or MSDS: Please include either the CAS (Chemical Abstract Service) number of each material (fill in the blank with the number), or state "MSDS" and append a copy to this document.
- 3. Annual quantity/units & Density/units: Please specify the amount of material consumed annually. Please use the units of mass-per-year (e.g., kg/yr, lb/yr). If you specify units of volume in lieu of mass, please provide the density.
- 4. Data quality indicators: See the Data Quality Indicators Index on p. iii for abbreviations. Please supply the DQI for the annual quantity value given.
- 5. Recycled content: Please specify the recycled content of each material identified. For example, 60/40/0 would represent a material that has 60% virgin material, 40% pre-consumer recycled and 0% post-consumer recycled content. Enter N/A (not applicable) for all components that are assemblies.
- 6. <u>Transportation information</u>: See the Transportation Modes Index on p. iii for mode abbreviations. Please specify where the material is coming from (location) and the number of trips made to your facility on an annual basis. % capacity represents what percent of the transport vehicle's total *load* was carrying the materials of interest.

Table 3a.	CAS#	Annual	Units	Density ³	Units	DQI	Recycled	Transportation	Informa	tion (Recei	iving) ⁶
Primary Materials ¹	or MSDS ²	Quantity ³				4	Content ⁵	Location	Mode	# trips	% cap.
1.											
2.									1		
3								1			
4											
									· · · ·		
0.					·						
		······	L		l			i			L
Primary material comments:											
											· · · ·
L											
											and the second se
Table 3b.	CAS#	Annual	Units	Density ³	Units	DQI	Recycled	Transportation]	nforma	ion (Recei	ving) ⁶
Table 3b. Ancillary Materials ¹	CAS # or MSDS ²	Annual Quantity ³	Units	Density ³	Units	DQI 4	Recycled Content ⁵	Transportation Location	nforma Mode	lion (Recei # trips	ving) ⁶ % cap.
Table 3b. Ancillary Materials ¹	CAS # or MSDS ²	Annuał Quantity ³	Units	Density ³	Units	DQI 4	Recycled Content ⁵	Transportation Location	Informa Mode	lion (Recei # trips	ving) ⁶ % cap.
Table 3b. Ancillary Materials ¹ 1. 2.	CAS # or MSDS ²	Annuał Quantity ³	Units	Density ³	Units	DQI 4	Recycled Content ⁵	Transportation	nforma Mode	iion (Recei # trips	ving) ⁶ % cap.
Table 3b. Ancillary Materials ¹ 1. 2. 3.	CAS # or MSDS ²	Annusł Quantity ³	Units	Density ³	Units	DQI 4	Recycled Content ⁵	Transportation Location	Mode	iion (Recei # trips	ving) ⁶ % cap.
Ancillary Materials ¹ 1. 2. 3. 4.	CAS # or MSDS ²	Annusł Quantity ³	Units	Density ³	Units	DQI 4	Recycled Content ⁵	Transportation Location	Mode	iion (Recei # trips	ving) ⁶ % cap.
Cancillary Materials ¹ 1. 2. 3. 4. 5.	CAS # or MSDS ²	Annusł Quantity ³	Units	Density ³	Units	DQI 4	Recycled Content ⁵	Transportation Location	Mode	ion (Recei # trips	ving) ⁶ % cap.
Ancillary Materials ¹ 1. 2. 3. 4. 5. 6.	CAS # or MSDS ²	Annusł Quantity ³	Units	Density ³	Units	DQI 4	Recycled Content ⁵	Transportation Location	nformat Mode	ion (Recei # trips	ving) ⁶ % cap.
Ancillary Materials ¹ 1. 2. 3. 4. 5. 6. 7.	CAS # or MSDS ²	Annusł Quantity ³	Units	Density ³	Units	DQI 4	Recycled Content ⁵	Transportation Location	Information Mode	ion (Recei # trips	ving) ⁶ % cap.
Ancillary Materials ¹ 1. 2. 3. 4. 5. 6. 7.	CAS # or MSDS ²	Annusł Quantity ³	Units	Density ³	Units	DQI 4	Recycled Content ⁵	Transportation Location	nforma Mode	ion (Recei # trips	ving) ⁶ % cap.
Table 3b. Ancillary Materials ¹ 1.	CAS # or MSDS ²	Annusł Quantity ³	Units	Density ³	Units	DQI 4	Recycled Content ⁵	Transportation Location	nforma Mode	ion (Recei # trips	ving) ⁶ % cap.
Table 3b. Ancillary Materials ¹ 1	CAS # or MSDS ²	Annual Quantity ³	Units	Density ³	Units		Recycled Content ⁵	Transportation Location	nformat Mode	ion (Recei # trips	ving) ⁶ % cap.

4. UTILITY INPUTS

- 1. <u>Annual quantity/units</u>: Please specify the amount of each utility consumed annually. If possible, please exclude nonprocess-related consumption. If not possible, please include a comment that nonprocess-related consumption is included.
- 2. Data quality indicators: See the Data Quality Indicators Index on p. iii for abbreviations. Please supply the DQI for the annual quantity value given.
- 3. <u>Transportation information</u>: See the Transportation Modes Index on p. iii for mode abbreviations. Please specify where the fuel is coming from (Location) and the number of trips made to your facility on an annual basis. Percent capacity represents what percent of the transport vehicle's total *load* was carrying the fuel of interest.
- 4. Individual Utility Notes:

Electricity:

The quantity of electricity should reflect only that used toward manufacturing the product of interest (identified on p. 2). One approach would be to start with your facility's total annual electrical energy consumption, estimate and remove nonprocess-related consumption, then estimate what portion of the remaining consumption is related to the specific operations of interest (if you manufacture more than one product). Please include consumption in all systems that use electricity for process-related purposes. Some examples include compressed air, chilled water, water deionization and HVAC consumption where clean or controlled environments are utilized.

Natural gas and LNG:

Please exclude all use for space heating or other nonprocess-related uses. If you choose to use units other than MCF (thousand cubic feet), please utilize only units of energy content or volume (e.g., mmBTU, therm, CCF).

Fuel oils:

Please use units of either volume or energy content (e.g., liters, cubic meters, mmBTU, MJ). Additionally, if the fuel oil is delivered by pipeline, enter "pipeline" in the Transportation Information space; if not delivered by pipeline, please include the associated transportation information.

All waters (e.g., deionized, city):

Please include all waters received onsite for process-related uses. Please indicate consumption in units of mass or volume.

Table 4.	Annual	Units	DQI ²	Transportation In	formation	(Receiving) ³	
Utilities ⁴	Quantity ¹			Location (City, State)	Mode	# trips	% сяр.
1. Electricity		мј					
2. Natural gas		MCF				د. هوشد که توجه در آن برد کرد.	num , in cost of st. or a
3. Liquified natural gas (LNG)		MCF					
4. Fuel oil - type #2 (includes distillate and diesel)		liters					
5. Fuel oil - type #4	· ·	liters					
6. Fuel oil - type #6 (includes residual)	·	liters		· · · · · · · · · · · · · · · · · · ·			
7. Other petroleum-based fuel		liters					
8. Water		liters		a la participante de la construcción de la construcción de la construcción de la construcción de la construcción Construcción de la construcción de l	in a star Anna an	and the second secon	ا المراجع بالمارين. مرکز فریک کرد المراجع
9.							
10.			· · · ·				
11.							
12.							
13.							
Utility comments:		•					
		,					
Utility comments:	. L	······		J	<u>}</u> .		-

5. PRODUCT OUTPUTS

1. Product Outputs: Product outputs are defined as useable products, materials, components or sub-assemblies.

- 2. CAS # or MSDS (if applicable): Please include either the CAS (Chemical Abstract Service) number of each material (fill in the blank with the number), or state "MSDS" and append a copy to this document.
- 3. <u>Annual quantity/units</u> & <u>Density/units</u>: Please specify the amount of material produced annually. Please use the units of mass-per-year (e.g., kg/yr, lb/yr). If you specify units of volume in lieu of mass, please provide the density.
- 4. Data quality indicators: See the Data Quality Indicators Index on p. iii for abbreviations. Please supply the DQI for the annual quantity value given.
- 5. <u>Recycled content (if known)</u>: Please specify the recycled content of each material identified. For example, 60/40/0 would represent a material that has 60% virgin material, 40% pre-consumer recycled and 0% post-consumer recycled content. Enter N/A (not applicable) for all components that are assemblies.
- 6. <u>Transportation information</u>: See the Transportation Modes Index on p. iii for mode abbreviations. Please specify where the material is being sent (location) and the number of trips made on an annual basis. % capacity represents what percent of the transport vehicle's total *load* was carrying the materials of interest.

Table 5.	CAS#	Annual	Units	Density ³	Units	DQI	Recycled	Transportation Information (Shipping			ping) ⁶
Product Outputs	or MSDS ²	Quantity ³				4	Content ⁵	Location	Mode	# trips	% cap.
1.							·				
2.											
3.											•
4		,									
5.				· · ·				· · · ·		k/k	
6	· · ·						· · · ·				
7								······································			
Product output comments:	1		<u>t</u>	1		L		L			L

6. AIR EMISSIONS

1. Air emissions: The emissions listed in the table below are some of the more common ones found in air release inventories; if you have information on other specific emissions, please include that information in the space provided. If you have any recent reporting forms or other air emission records, please attach copies to this questionnaire. Also, if you have information on stack* as well as fugitive* emissions, please copy this page and place each set of emissions on a different page. The energy consumed in any equipment used onsite to treat air emissions should be included in the utilities values on p. 4.

Stack emissions* are releases to air that occur through confined air streams, such as stacks, vents, ducts, or pipes.

Fugitive emissions* are all releases to air that are not released through a confined air stream. Fugitive emissions include equipment leaks, evaporative losses from surface

impoundments and spills, and releases from building ventilation systems.

and a second and a second contract of the second
2. Annual quantity/units: Please specify the amount of air emissions generated annually. Please use units of mass-per-year (e.g., kg/yr, lb/yr).

3. Data quality indicators: See the Data Quality Indicators Index on p. iii for abbreviations. Please supply the DQI for the annual quantity value given.

Table 6.	CAS	Annual	Units	DQI	Table 6 (continued).	CAS	Annual	Units	DQL
Air Emissions ¹	number	Quantity ²		3	Air Emissions ¹	number	Quantity ²		3
Total particulates					Ammonia	7664-41-7			
Particulates < 10 microns (PM-10)					Arsenic	7440-38-2			
Sulfur oxides (SOx)					Chromium	7440-47-3			
Nitrogen oxides (NOx)					Copper	7440-50-8			
Carbon monoxide	630-08-0				Lead	7439-92-1			
Carbon dioxide	124-38-9				Manganese	7439-96-5			
Methane	74-82-8				Мегсигу	7439-98-7			
Benzene	71-43-2				Nickel	7440-02-0			
Toluene	108-88-3		<u> </u>		Other emissions:	a deres and decars being	4.1426-P# 14. 44	i si kasali i	
Xylenes	1330-20-7	·			1.				·
Naphthalene	91-20-3				2.				
Total nonmethane VOCs					3.	<u> </u>			
Other speciated hydrocarbon emissions:			te solitesta	a an	4.		-	ļ	1
1.		l			5.				
2.		<u> </u>			6.				
3.					7.				
4.					8.				L
5.					Air emission comments:				
6.									
7.									
8.								-	
9.									
10.									

7. WASTEWATER RELEASES & CONSTITUENTS

- 1. Annual quantity/units: Please specify the amount of wastewater(s) generated annually. Please use units of mass-per-year (e.g., kg/yr, lb/yr). If multiple streams exist, please copy this page and fill it out for each stream.
- 2. Data quality indicators: See the Data Quality Indicators Index on p. iii for abbreviations. Please include one DQI for the annual wastewater stream quantity value supplied, and one DQI for the wastewater constituents information supplied. If more than one DQI is applicable to the wastewater constituents data, please clarify this in the comment section.
- 3. Wastewater constituents: Please let us know what type of values you are supplying (e.g., daily maximums, monthly averages, annual averages). Additionally, if you have any recent reporting forms or other wastewater constituent records, please attach them to this questionnaire. The energy consumed in any equipment used onsite to treat wastewater releases should be included in the utilities values on p. 4.

4. Concentration/units: Please specify the concentration of wastewater constituents generated annually. Please utilize the units of mass-per-volume (e.g., mg/liter, lb/gal).

5. Wastewater treatment/disposal (WW T/D) method: See the Wastewater Treatment/Disposal Methods Index on p. iii for method abbreviations.

Table 7s.	Annual	Units	DQI for Wastewater Annual Quantity ²	DQI for Wastewater
Wastewater Stream	Quantity ¹			Constituents ²
				-
	L			

Table 7b.	CAS	Concen-	Units	WW T/D	Table 6b (continued).	CAS	Concen-	Units	WW T/D
	number	tration		Method		number	tration		Method
Wastewater Constituents					Wastewater Constituents	· · · · · · · · · · · · · · · · · · ·		ļ	
Dissolved solids		····	· .		Mercury	7439-98-7			
Suspended solids					Lead	7439-92-1			
Chemical Oxygen Demand (COD)					Nitrogen	7727-37-9	·		
Biological Oxygen Demand (BOD)	******				Zinc	7440-66-6			
Oil & grease	*****	·			Tin	7440-31-5			
Hydrochloric acid	7647-01-0				Ferrous sulfate	7720-78-7	·		
Sulfuric acid	7664-93-9			·	Ammonia	7664-41-7			
Other acids (please specify):	a the state of a second	hi kaca ta Unero, ca I	ed to de take the		Nitrates	***			
1.					Pesticides				
2.					Other constituents:			Security States	Same in the
Phosphorus	7723-14-0		<u> </u>		1.				
Phosphates			<u> </u>		2.				
Sulfates					3.				
Fluorides	-+				4.				
Cyanide					5.				
Chloride					6.				
Chromium	7440-47-3				Wastewater comments:				
Iron	7439-89-6				•				
Aluminum	7429-90-5				,				
Nickel	7440-02-0				s				

8. HAZARDOUS & NONHAZARDOUS WASTES

- 1. Hazardous wastes and EPA hazardous waste numbers: Please list your waste streams that are considered hazardous by the U.S. EPA. Include the hazardous waste codes for any hazardous waste you include.
- 2. Annual quantity/units & Density/units: Please specify the amount of waste generated annually. Use units of mass-per-year (e.g., kg/yr, lb/yr). Please also provide the density for each waste.
- 3. Data quality indicators: See the Data Quality Indicators Index on p. iii for abbreviations. Please supply the DQI for the annual quantity value given.
- 4. Management method: See the Management Methods Index on p. iii for abbreviations. If none are applicable, please indicate other and use the comments section to expound.
- 5. <u>Transportation information</u>: See the Transportation Modes Index on p. iii for abbreviations. Please specify where the waste is sent (location) and the number of trips made from your facility on an annual basis. % capacity represents what percent of the transport vehicle's total *load* was carrying the waste of interest.

Table 8a.	EPA Haz.	Annual	Units	Density ²	Units	DQI	Mgmt.	On or	Transportation Inl	formatic	on (Shipp	ing) ⁵
Hazardous Wastes ¹	Waste # ¹	Quantity ²				3	method ⁴	offsite?	Location	Mode	# trips	% cap.
EXAMPLE: Spent solvent (toluene)	F005	20,000	kg/yr	0.9	kg/liter	M	i le	off	Indianapolis, IN	7	24	40
1.									······			
2.												
3	· [
	· · · · · · · · · · · · · · · · · · ·							· · ·	· · · · · · · · · · · · · · · · · · ·	· · · · · ·		
3.												
<u>6.</u>		l		[
7.	l					·						· · · · · · · · · · · · · · · · · · ·
8.	<u> </u>	I		1		L		L	[<u> </u>
Hazardous waste comments:												
									•			
Table 8b.		Annual	Units	Density ²	Units	DQI	Mgmt.	On or	Transportation Inf	ormatio	on (Shinr	5
Nonhererdous Wester										Of many		ung)
Tionnazai dous vrastes		Quantity ²				3	method4	offsite?	Location	Mode	# trips	ing) % cap.
EXAMPLE: Waste metal chips		Quantity ² 22,000	kg/yr	1,000	kg/m3	, С	method ⁴	offsite?	Location Scottsdale, AZ	Mode	# trips	ng) % cap. 100
EXAMPLE: Waste metal chips		Quantity ² 22,000	kg/yr	1,000	kg/m3	3 	method ⁴	offsite? off	Location Scottsdale, AZ	Mode	# trips	ng) % cap. 100
EXAMPLE: Waste metal chips 1. 2.		Quantity ²	kg/yr	1,000	kg/m3	3 	method ⁴	offsite?	Location Scottsdale, AZ	Mode	# trips	мд) % сяр. 100
EXAMPLE: Waste metal chips		Quantity ² 22,000	kg/yr	1,000	kg/m3	3 C	method ⁴	offsite?	Location Scottsdale, AZ	Mode	# trips	пад) % сяр. 100
EXAMPLE: Waste metal chips		Quantity ² 22,000	kg/yr	1,000	kg/m3	3 C	R R	offsite?	Location Scottsdale, AZ	Mode	# trips	мд) % сяр. 700
EXAMPLE: Waste metal chips		Quantity ² 22,000	kg/yr	1,000	kg/m3	3 C	R R	offsite?	Location Scottsdale, AZ	Mode	# trips	"100 "100
EXAMPLE: Waste metal chips 1. 2. 3. 4. 5. 6.		Quantity ² 22,000	kg/yr	<u>1,000</u>	kg/m3	3 	R R	offsite?	Location Scottsdale, AZ	Mode	# trips	лду) % сяр. 100
EXAMPLE: Waste metal chips 1. . 2. . 3. . 4. . 5. . 6. . 7 .		Quantity ² 22,000	kg/yr	1,000	kg/m3	3 <u>C</u>	method ⁴	offsite?	Location Scottsdale, AZ	Mode	# trips	11ng) % cap. 700
EXAMPLE: Waste metal chips 1. 2. 3. 4. 5. 6. 7.		Quantity ² 22,000	<u>kg/yr</u>	1,000	kg/m3	3 <u>C</u>	Rethod	offsite?	Location Scottsdale, AZ	Mode	# trips	111g) % cap. 700
EXAMPLE: Waste metal chips		Quantity ² 22,000	kg/yr	<u>1,000</u>	kg/m3	3 C	R House	offsite?	Location Scottsdale, AZ	Mode	# trips 2	108) % cap. 700
EXAMPLE: Waste metal chips 1. 2. 3. 4. 5. 6. 7. Nonhazardous waste comments:		Quantity ² 22,000	kg/yr	1,000	kg/m3	3 <u>C</u>	R R	offsite?	Location Scottsdale, AZ	Mode	# trips	% cap. 700

this page intentionally left blank

APPENDIX J

LIFE-CYCLE INVENTORY TABLES

this page intentionally left blank

APPENDIX J

LIFE-CYCLE INVENTORY TABLES

Table J-1	. CRT prima	ry materia	ls (kg/fun	ctional unit	:)	
Material	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
ABS resin	0	4.24E-01	0	0	4.24E-01	0.0649%
Aluminum (elemental)	0	3.60E-01	0	0	3.60E-01	0.0552%
Amyl acetate (mixed isomers)	0	1.20E-03	0	0	1.20E-03	0.0002%
Aquadag	0	2.06E-02	0	0	2.06E-02	0.0032%
Assembled CRT monitor	0	0	2.20E+01	0	2.20E+01	3.3765%
Audio cable assembly	0	9.45E-02	0	0	9.45E-02	0.0145%
Barium Carbonate	0	2.97E-01	0	0	2.97E-01	0.0455%
Bauxite (Al2O3, ore)	1.37E+00	0	0	0	1.37E+00	0.2096%
Blue Phosphor (ZnS)	0	3.84E-03	0	0	3.84E-03	0.0006%
Blue Phosphor (ZnS.Ag.Al)	0	1.67E-03	0	0	1.67E-03	0.0003%
Borax	0	8.00E-03	0	0	8.00E-03	0.0012%
Cables/wires	0	3.94E-01	0	0	3.94E-01	0.0604%
Cathode ray tube (CRT)	0	1.07E+01	0	0	1.07E+01	1.6381%
Coal. average (in ground)	3.57E+00	5.15E+00	1.79E+02	1.79E-02	1.88E+02	28.8170%
Connector	0	5.67E-02	0	0	5.67E-02	0.0087%
CRT glass unspecified	0	9 76E+00	0	0	9 76E+00	1 4950%
CRT magnet assembly	0	7 56E-02	0	0	7 56E-02	0.0116%
CRT shield assembly - ASTM A366/CC#2	0	2 42E-01	0	0	2 42E-01	0.0370%
Deflection Voke assembly	0	1.51E-01	0	0	1.51E-01	0.0232%
Demagnetic coil - PU coated paper	0	1.31E 01	0	0	1.31E 01	0.0193%
Flectron gun	0	1.20E 01	0	0	1.20E 01	0.0154%
Ferrite	0	1.70E-01	0	0	1.01E 01	0.0261%
Frit	0	6.67E-02	0	0	6.67E-02	0.0201%
Fuel oil #4	0	0.0712 02	0	-9 53E-02	-9 53E-02	-0.0146%
Glass unspecified	0	4.91E-02	0	0	4.91E-02	0.0075%
Green Phoenhor (7nS)	0	3.34E.03	0	0	3.34E.03	0.0005%
Green Phosphor (ZnS (u Al)	0	1.34E-03	0	0	1.34E.03	0.0003%
Iron (Eq. org)	6 00E±00	1.54L-05	0	0	6 00E±00	1.0567%
Iron ere	2.27E 01	0	0	0	0.90E+00	0.02629/
	2.37E-01	0	0	0	2.37E-01	0.14409/
Land	9.40E-01	4 94E 01	0	0	9.40E-01	0.1449/0
Lead (Ph. ara)	4 06E 01	4.9412-01	0	0	4.94E-01	0.0750%
Natural gas	4.90E-01	1 47E+00	1 40E±01	8 88E 02	4.90E-01	2 3551%
Natural gas	8 41E 01	2.27E+00	1.401	-8.88E-02	2.47E+00	0.27869/
Niakal (Ni, ara)	0.41E-01	3.27E+00	0	-1.04E+00	2.47E+00	0.378076
Niekel Alley (inver)	9.801-02	2 72E 01	0	0	9.80E-02	0.011078
Detroloum (in ground)	1 22E+00	2.72E-01	2 805 100	1.52E+00	2.72E-01	57 50160/
Phoenhote estan	1.52E+00	5.72E∓02 8.21E_02	3.80E+00	-1.32E+00	5.73E∓02	0.00120/
Phosphate ester	0	8.51E-05	0	0	0.31E-03	0.0013%
Polycarbonate resin	0	9.23E-01	0	0	9.23E-01	0.1413%
Polystyrene (PS, high impact)	0	1.51E-01	0	0	1.51E-01	0.0232%
Potassium Cardonale	0	5./8E-01	0	0	5./8E-01	0.05/9%
POWER COLU ASSEILIDIY	0	1.13E-01	0	0	1.13E-01	0.01/4%
rre	0	7.35E-01	0	0	7.35E-01	0.1126%
PWD lawingto	0	8.4/E-01	0	0	8.4/E-01	0.1298%
r w D-iaminate	U	8.4/E-01	U	0	8.4/E-01	0.1298%

		ary materia	is (ng/iui			
Material	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Recycled CRT Glass	0	2.06E+00	0	0	2.06E+00	0.3155%
Red Phosphor (Y2O2S)	0	4.65E-03	0	0	4.65E-03	0.0007%
Red Phosphor (Y2O2S.Eu)	0	1.33E-03	0	0	1.33E-03	0.0002%
Sand	0	2.40E+00	0	0	2.40E+00	0.3678%
Sand (in ground)	2.42E-05	0	0	0	2.42E-05	3.71E-08
Silica	0	5.33E-03	0	0	5.33E-03	0.0008%
Sodium Carbonate	0	4.88E-01	0	0	4.88E-01	0.0747%
Sodium chloride (NaCl, in ground or in sea)	3.17E-04	0	0	0	3.17E-04	4.85E-07
Solder (63% tin; 37% lead)	0	5.08E-02	0	0	5.08E-02	0.0078%
Solder, unspecified	0	2.67E-02	0	0	2.67E-02	0.0041%
Steel	2.48E-06	5.16E+00	0	0	5.16E+00	0.7907%
Strontium Carbonate	0	3.31E-01	0	0	3.31E-01	0.0508%
Styrene-butadiene copolymers	0	8.27E-01	0	0	8.27E-01	0.1268%
Sulfur	1.03E-05	0	0	0	1.03E-05	1.58E-08
Tricresyl phosphate	0	2.30E-02	0	0	2.30E-02	0.0035%
Triphenyl phosphate	0	5.29E-02	0	0	5.29E-02	0.0081%
Uranium, yellowcake	0	3.81E-04	4.85E-03	4.85E-07	5.23E-03	0.0008%
Video cable assembly	0	1.13E-01	0	0	1.13E-01	0.0174%
Zircon Sand	0	5.43E-02	0	0	5.43E-02	0.0083%
Total Primary Inp	uts 1.58E+01	4.21E+02	2.19E+02	-3.32E+00	6.53e+02	100.00%

Table J-1.	CRT DI	·imarv	materials	(kg/	functional	unit)
		JIII	materials	(11 5/	lunctiona	ame

2,4,4 fmichtylpentane01,5%-04001,5%-06Aceine03,17e-03003,17e-031,60e.05Acylic holyner, unspecified07,72e-02009,13e-034,61e-04Alkali sock (no textilize acid waste water)05,75e-02005,75e-022,75e-03Aluminum Oxide03,74e-030003,74e-033,74e-03Annonian01,19e-04001,19e-046,03e-04Annonian01,19e-04007,76e-023,29e-03Annonian Dirboride02,04e-03003,50e-053,50e-05Annonian Undrode08,91e-040003,50e-05Annonian Wadrede08,92e-050003,56e-05Annonian Wadrede08,92e-050003,56e-05Annonian Wadrede08,92e-050001,56e-05Annonian Wadrede08,92e-050001,56e-05Annonian Wadrede08,92e-050001,56e-05Barinan Saltate (BSO4, inground)1,10e-034,47e-02001,86e-05Barinan Saltate (BSO4, inground)1,10e-034,77e-020001,27e-04Barinan Saltate (ADO, ore)1,10e-034,77e-030001,27e-04Barinan Saltate (ADO, ore)1,10e-034,77e-030 <t< th=""><th>Material</th><th>Upstream</th><th>Mfg</th><th>Use</th><th>EOL</th><th>Total</th><th>% or Fraction of Total</th></t<>	Material	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Acetone 0 3.17e-04 0 0 9.13e-03 100 9.13e-03 4.61e-04 Araly far Polymer, unspecified 0 9.72e-02 0 0 9.72e-02 3.90e-03 Alkal solat (to neutralize acid waste water) 0 5.45e-02 0 0 3.77e-02 1.70e-03 Aluminum Suffar 7.41e-04 0 0 0 1.73e-03 3.77e-02 0 0 3.77e-02 1.95e-04 6.638e-06 Ammoninu Suffar 0 2.19e-04 0 0 1.19e-04 6.638e-06 Ammoninu birhoride 0 7.75e-02 0 0 7.97e-02 3.98e-03 Ammoninu Thoride 0 7.75e-02 0 0 7.97e-02 3.98e-03 Ammoninu Thoride 0 8.91e-04 0 0 8.91e-04 1.68e-05 Ammoninu Thoride 0 8.92e-05 1.38e-04 1.68e-04 1.68e-05 1.38e-04 Ammoninu Thoride 0 3.16e-04 0 0 6.	2,2,4-trimethylpentane	0	1.50e-04	0	0	1.50e-04	7.59e-06
Acy Lie Dymer, unspecified 0 9.13e-03 0 0 9.13e-03 4.61e-04 Akali sodu (tro neutralize acid waste water) 0 5.45e-02 0 00 7.72e-02 3.90e-03 Aluminum Stride 7.41e-04 0 0 0 7.71e-04 3.90e-03 Aluminum Stride 7.41e-04 0 0 0 7.71e-02 3.90e-03 Ammonium Difluoride 0 7.76e-02 0 0 7.76e-02 3.92e-03 Ammonium Dichtore 0 3.50e-05 0 0 8.91e-04 4.50e-05 Ammonium Dichtore 0 3.16e-04 0 0 8.92e-05 4.50e-05 Ammonium Datate 0.89e-05 0 0 0 1.10e-03 5.58e-05 Bariam sulfate 2.66e-03 0 0 0 1.16e-04 3.18e-05 Bariam sulfate 6.31e-04 1.0e-03 4.74e-02 0 0 4.73e-02 2.31e-03 Bariam sulfate 6.31e-04 1.0e-04	Acetone	0	3.17e-04	0	0	3.17e-04	1.60e-05
Akali clasning agent 0 7.72-02 0 0 5.45e-02 3.90e-03 Akali soda (to neutralize acid waste water) 0 5.45e-02 0 0 3.37e-02 1.70e-03 Aluminum Oxida 7.41e-04 0 0 0 1.19e-04 6.03e-06 Aumonia 0 1.19e-04 0 0 1.19e-04 6.03e-06 Anmonium Diribuoride 0 7.76e-02 0 0 3.5ee-05 1.73e-04 Ammonium Diribuoride 0 8.91e-04 0 0 3.5ee-05 1.76e-06 Ammonium Okatate 0 8.91e-04 0 0 7.90e-02 3.90e-03 4.50e-05 Ammonium Oxatate 0 8.92e-05 0 0 8.91e-04 4.50e-05 Ammonium Oxatate 0 8.92e-05 0 0 2.86e-03 1.60e-05 Barima suffare 0.631e-04 3.0e-04 0 0 2.86e-03 1.86e-04 Barima suffare 0.631e-04 3.16e-04 3.18e-05 1.35e-04 1.86e-05 1.27e-04 4.57e-02 2.31e-03	Acrylic Polymer, unspecified	0	9.13e-03	0	0	9.13e-03	4.61e-04
Akali sol (to neutralize acid waste water) 0 5.45-02 2.75-03 Aluminum Nxide 0 3.37-02 0 0 3.37-02 Aluminum Stafae 7.41-e14 0 0 0 7.41-e34 3.74-e45 Ammonium Stafae 0 1.19-e94 0 0 2.04e-03 0.0 2.04e-03 0.0 2.04e-03 0.0 2.04e-03 0.0 3.92-e03 0.0 7.76-02 3.92-e03 0.0 0 8.91-e04 4.50e-05 0.0 8.91-e04 4.50e-05 0.0 8.91-e04 4.50e-05 5.5ee-05 5.5ee-05 5.5ee-05 5.5ee-05 5.5ee-05 5.5ee-05 5.5ee-05 5.5ee-05 5.5ee-05 5.20e-07 0 0 0 4.37e-02 2.31e-03 5.5ee-05 5.5ee-05 5.5ee-05 5.5ee-05 5.5ee-05 5.20e-07 2.62e-0	Alkali cleaning agent	0	7.72e-02	0	0	7.72e-02	3.90e-03
Aluminum Oxide 0 3.37e-02 0 0 3.37e-02 1.70e-04 Aluminum suffate 7.41e-04 0 0 7.41e-04 3.74e-05 Alumonium bifluoride 0 2.04e-03 0 0 2.04e-03 1.03e-04 Ammonium bifluoride 0 7.76e-02 0 0 3.50e-05 0 0 3.50e-05 1.76e-02 Ammonium Dichromate 0 8.91e-04 0 8.91e-04 4.50e-05 Ammonium floride 0 8.92e-05 0 0 7.99e-02 3.99e-03 Ammonium Oxalate 0 8.92e-05 0 0 8.92e-05 4.50e-06 Ammonium Oxalate Monolydrate 2.68e-03 0 0 0 1.60e-05 5.59e-05 Barrium suffare (DaSO4, in ground) 1.10e-03 4.47e-02 0 -1.14e-04 4.57e-02 2.31e-03 Barrium suffare (DaSO4, orcy) 1.10e-03 4.47e-02 0 -1.14e-04 4.57e-02 2.31e-03 Barrium suffare (DaSO4, orcy)	Alkali soda (to neutralize acid waste water)	0	5.45e-02	0	0	5.45e-02	2.75e-03
Aluminum sulfae 7.41e-04 0 0 7.41e-04 3.74e-05 Anmonium bifuoride 0 2.04e-03 0 0 1.19e-04 6.03e-06 Ammonium bifuoride 0 2.04e-03 0 0 7.76e-02 3.92e-05 Ammonium Induride 0 8.91e-04 0 8.91e-04 4.50e-05 Ammonium Induride 0 8.92e-05 0 0 7.90e-02 3.99e-05 Ammonium Induride 0 8.92e-05 0 0 8.91e-04 4.50e-05 Ammonium Oxalate 0 3.16e-04 0 0 8.16e-04 1.60e-05 Barium sulfac (BaSO4, in groun) 1.10e-03 0 0 0 2.68e-03 1.38e-04 Barium sulfac (BaSO4, in groun) 1.10e-03 4.47e-02 0 0 4.38e-03 2.30e-07 Barium sulfac (BaSO4, in groun) 2.08e-07 0 0 0 2.32e-03 2.30e-04 Calcium Sulfac (DaSO4, orc) 2.30e-07 0 0 4.32e-03	Aluminum Oxide	0	3.37e-02	0	0	3.37e-02	1.70e-03
Ammonia01.19e.04001.19e.046.03e.04Ammonium bifloride02.04e.03002.04e.031.03e.04Ammonium Lobride03.50e.05003.50e.051.76e.02Ammonium Dichromate08.91e.0408.91e.044.50e.05Ammonium Invoride08.91e.04008.92e.053.99e.03Ammonium Lokate08.92e.05008.92e.054.50e.05Ammonium Calate Monolydrate03.16e.040001.16e.03Barium sulfare2.68e.0300001.55e.05Baruxite6.31e.040006.31e.041.38e.05Baruxite6.31e.040002.48e.031.25e.04Baruxite09.54e.02002.48e.031.25e.04Boria caid04.73e.03004.73e.032.30e.04Calcium Mydroxide09.54e.02002.48e.032.30e.04Calcium Sulfate (CSOC4, ore)2.03e.050002.30e.05Calcium Sulfate (CSOC4, ore)1.09e.070002.32e.031.68e.04Calcium Mydroxide09.54e.02002.48e.031.68e.04Calcium Sulfate (CSOC4, ore)000002.30e.05Calcium Sulfate (CSOC4, ore)000000Calcium Myd	Aluminum sulfate	7.41e-04	0	0	0	7.41e-04	3.74e-05
Ammonium biluoride 0 2.04-03 0 0 2.04-03 Ammonium bichromate 0 3.50-05 0 0 3.50-05 Ammonium Dichromate 0 8.91-04 0 0 8.91-04 4.500-05 Ammonium Nydroxide 0 8.92-05 0 0 8.92-05 4.500-05 Ammonium Oxalate 0 8.92-05 0 0 8.92-05 4.500-05 Ammonium Oxalate 0 8.92-05 0 0 8.92-05 4.500-05 Ammonium Oxalate 0 3.160-04 0 0 3.160-04 1.00-05 Baruin sulfate (BaSO4, in ground) 1.10e-03 0 0 0 6.31e-04 3.18e-05 Bauxite (102,0,cry) 1.10e-03 4.47e-02 -1.14e-04 4.47e-02 2.31e-03 Boria acid 0 4.47e-02 0 0 2.48e-03 1.25e-04 Boria acid 0 4.47e-02 0 0 2.48e-03 1.25e-04 Boria acid 0 4.47e-02 0 0 2.42e-03 1.25e-04 <td>Ammonia</td> <td>0</td> <td>1.19e-04</td> <td>0</td> <td>0</td> <td>1.19e-04</td> <td>6.03e-06</td>	Ammonia	0	1.19e-04	0	0	1.19e-04	6.03e-06
Ammonium Libride 0 7,76-02 0 0 7,76-02 3,50e-05 0 0 3,50e-05 1,76e-06 Ammonium Dichromate 0 8,91e-04 0 0 8,91e-04 4,50e-05 Ammonium Oxalate 0 8,92e-05 0 0 8,92e-05 4,50e-05 Ammonium Oxalate 0 8,92e-05 0 0 1,6e-04 1,6e-04 Ammonium Sulfate 2,86e-03 0 0 0 2,68e-03 1,35e-04 Barium sulfate 2,86e-03 4,47e-02 0 -1,14e-04 4,57e-02 2,31e-03 Bauxite 6,61e-04 0 0 0 2,48e-03 1,25e-04 Borax 5,20e-07 0 0 0 2,48e-03 2,39e-04 Calcium fydroxide 0 9,24e-02 0 0 4,73e-03 2,39e-04 Calcium fydroxide 0 9,24e-02 0 0 2,30e-05 3,07e-06 Calcium fydroxide 0 9,24e	Ammonium bifluoride	0	2.04e-03	0	0	2.04e-03	1.03e-04
Ammonium Dichromate 0 3.50e-05 0 0 3.50e-05 Ammonium fluoride 0 7.90e-02 0 0 3.90e-03 Ammonium Oxalate 0 8.92e-05 0 0 8.92e-05 1.00e-03 Ammonium Oxalate Monohydrate 0 3.16e-04 0.00 0 2.68e-03 1.13e-04 Barium sulfate (BaSO4, in ground) 1.10e-03 0 0 0 6.51e-04 3.16e-04 3.16e-04 3.18e-05 Bauxite (LOS, orcy) 1.10e-03 4.47e-02 -1.14e-04 4.57e-02 2.31e-03 Borax 5.20e-07 0 0 0 2.48e-03 1.25e-04 Borax 5.20e-07 0 0 0 2.48e-03 2.39e-03 Boria acid 0 4.47a-03 0 0 2.48e-03 2.39e-04 Calcium sulfate (aSO4, orcy) 2.30e-05 0 0 0 8.8e-03 3.06e-04 3.06e-04 3.06e-03 3.06e-03 3.06e-03 3.06e-03 3.06e-03	Ammonium chloride	0	7.76e-02	0	0	7.76e-02	3.92e-03
Anmonium Ihuoride 0 8.91=04 0 0 8.91=04 4.50e.05 Anmonium Nxalate 0 8.92e-05 0 0 8.92e-05 4.50e.06 Anmonium Oxalate 0 8.92e-05 0 0 8.92e-05 4.50e.06 Anmonium Oxalate Monohydrate 0 3.16e-04 0 0 2.16e-04 1.60e.05 Barium sulfate (BSO4, in ground) 1.10e-03 0 0 0 1.10e-03 5.55e-05 Bauxite (Al2O3, orc) 1.10e-03 4.47e-02 0 -1.11e-04 4.57e-02 2.21e-03 Barta (Al2O3, orc) 1.10e-03 4.47e-02 0 0 0 2.48e-03 1.25e-04 Bora acid 0 4.37e-03 0 0 4.73e-03 2.39e-04 Calcium fulfate (CaSO4, orc) 2.30e-05 0 0 0 2.27e-01 6.42e-03 Calcium sulfate 6.08e-05 0 0 0 2.30e-05 1.16e-06 Calcium sulfate (CaSO4, orc) 2.30e-05	Ammonium Dichromate	0	3.50e-05	0	0	3.50e-05	1.76e-06
Ammonium hydroxide 0 7.90e-02 0 0 7.90e-02 3.99e-03 Ammonium Oxalate 0 8.92e-05 0 0 8.92e-05 4.50e-06 Barium sulfate 2.68e-03 0 0 0 2.68e-03 1.15e-04 Barium sulfate 2.68e-03 0 0 0 0.631e-04 3.18e-05 Barium sulfate 6.31e-04 0 0 6.31e-04 3.18e-05 Bauxite 6.31e-04 0 0 6.31e-04 3.18e-05 Barusite (Al2O3, ore) 1.10e-03 4.47e-02 0 -1.14e-04 4.57e-02 2.31e-03 Berto acid 0 4.73e-03 0 0 2.8e-03 2.39e-04 Calcium Chloride 0 1.27e-01 0 0 1.25e-04 8.1e-03 Calcium sulfate 6.08e-05 0 0 0 2.39e-04 4.81e-03 Calcium sulfate 6.08e-05 0 0 0 2.39e-05 3.16e-04	Ammonium fluoride	0	8.91e-04	0	0	8.91e-04	4.50e-05
Ammonium Oxalate 0 8,92e-05 0 0 8,92e-05 4,50e-06 Ammonium Oxalate Monohydrate 0 3,16e-04 0 0 3,16e-04 1,60e-05 Barium sulfate (BaSO4, in ground) 1,10e-03 0 0 0 6,31e-04 3,55e-05 Bauxite (Al2O3, ore) 1,10e-03 4,47e-02 0 -1,14e-04 4,57e-02 2,31e-03 Bentonite (in ground) 2,48e-03 0 0 0 2,48e-03 1,25e-04 Boria caid 0 4,73e-03 0 0 4,73e-03 2,39e-04 Calcium Choride 0 1,27e-01 0 0 1,27e-01 6,42e-03 Calcium Sulfate 6,08e-05 0 0 0 2,30e-05 1,16e-06 Calcium sulfate 6,08e-05 0 0 0 2,30e-05 1,66e-05 Calcium sulfate 0 3,28e-03 0 0 2,30e-05 1,66e-05 Calcium sulfate 0,08e-05 2,28e-03 0	Ammonium hydroxide	0	7.90e-02	0	0	7.90e-02	3.99e-03
Ammonium Oxalate Monohydrate 0 3.16e-04 0 0 3.16e-04 1.60e-03 Barium sulfate 2.68e-03 0 0 0 1.35e-04 Barium sulfate 6.31e-04 0 0 0 6.31e-04 3.18e-05 Bauxite 6.31e-04 0 0 0 6.31e-04 3.18e-05 Bauxite (Al2O3, ore) 1.10e-03 4.47e-02 0 -1.14e-04 4.57e-02 2.31e-03 Bertonite (in ground) 2.48e-03 0 0 0 2.26e-07 0 0 0 2.26e-07 2.26e-07 Boria caid 0 4.73e-03 0 0 4.73e-03 2.39e-04 2.48e-03 2.39e-04 2.48e-03 2.36e-02 4.81e-03 Calcium sulfate 6.08e-05 0 0 0 2.38e-03 1.16e-06 Calcium sulfate 6.08e-05 3.07e-06 2.04e-03 1.66e-04 Calcium sulfate (CaSO4, ore) 2.30e-05 0 0 3.28e-06 1.66e-06 Chorine	Ammonium Oxalate	0	8.92e-05	0	0	8.92e-05	4.50e-06
Barium sulfate 2.68c-03 0 0 0 2.68c-03 1.35c-04 Barium sulfate (BaSO4, in ground) 1.10e-03 0 0 0 6.31c-04 3.18c-05 Bauxite (Al2O3, orc) 1.10e-03 4.47c-02 0 -1.14c-04 4.57c-02 2.31c-03 Bentonite (in ground) 2.48c-03 0 0 0 2.48c-03 1.25c-04 Boric acid 0 4.73c-03 0 0 4.73c-03 2.30c-04 Calcium Chloride 0 4.73c-03 0 0 4.73c-03 2.39c-04 Calcium sulfate 6.08c-05 0 0 0 2.39c-04 4.81c-03 Calcium sulfate (CaSO4, ore) 2.30c-05 0 0 0 2.30c-05 3.07c-06 Calcium sulfate (CaSO4, ore) 2.30c-05 0 0 0 2.30c-05 3.85c-06 Chronium Oxide 0 7.63c-05 0 0 7.63c-05 2.88c-06 2.83c-063 3.165c-04 Chronium Oxide 0 </td <td>Ammonium Oxalate Monohydrate</td> <td>0</td> <td>3.16e-04</td> <td>0</td> <td>0</td> <td>3.16e-04</td> <td>1.60e-05</td>	Ammonium Oxalate Monohydrate	0	3.16e-04	0	0	3.16e-04	1.60e-05
Barium sulfate (BaSO4, in ground) 1.10e-03 0 0 0 1.10e-03 5.55e-05 Bauxite (A2O3, ore) 1.10e-03 4.47e-02 0 -1.14e-04 4.57e-02 2.31e-03 Bentonite (in ground) 2.48e-03 0 0 0 2.48e-03 1.25e-04 Bora acid 0 4.73e-03 0 0 4.73e-03 2.39e-03 Calcium Chloride 0 1.27e-01 0 0 4.73e-03 2.39e-03 Calcium Sulfate (CaSO4, ore) 2.30e-05 0 0 0 5.4e-02 4.81e-03 Calcium Sulfate (CaSO4, ore) 2.30e-05 0 0 0 3.28e-03 0.60 0 4.03e-02 2.04e-03 2.04e-03 1.65e-04 0 0 2.04e-03 2.04e-03 2.04e-03 2.04e-03 2.04e-03 2.04e-03 2.04e-03 2.05e-05 2.85e-05	Barium sulfate	2.68e-03	0	0	0	2.68e-03	1.35e-04
Bauxite 6.31e-04 0 0 6.31e-04 3.18e-05 Bauxite (AI2O3, ore) 1.10e-03 4.47e-02 0 -1.14e-04 4.57e-02 2.31e-03 Bentonite (in ground) 2.48e-03 0 0 2.48e-03 1.25e-04 Borax 5.20e-07 0 0 0 4.73e-03 2.39e-04 Calcium Chloride 0 1.27e-01 0 0 4.73e-03 2.39e-04 Calcium Sulfate 6.08e-05 0 0 0 9.54e-02 4.81e-03 Calcium sulfate (CaSO4, ore) 2.30e-05 0 0 0 2.30e-05 1.16e-06 Calcium sulfate (CaSO4, ore) 2.30e-05 0 0 3.28e-03 1.6e-06 Chlorine 0 4.03e-02 0 0 3.28e-03 1.6e-06 Chlorine 0 3.28e-03 0 0 3.28e-03 1.6e-06 Chlorine 0 4.03e-02 0 0 3.85e-06 2.04e-03 Chromium Oxi	Barium sulfate (BaSO4, in ground)	1.10e-03	0	0	0	1.10e-03	5.55e-05
Bauxite (AI2O3, ore) 1.10e-03 4.47e-02 0 -1.14e-04 4.57e-02 2.31e-03 Bentonite (in ground) 2.48e-03 0 0 0.48e-03 1.25e-04 Borax 5.20e-07 0 0 0.473e-03 2.39e-04 Calcium Chloride 0 1.27e-01 0 0 1.27e-01 6.42e-03 Calcium Mydroxide 0 9.54e-02 0 0 9.54e-02 4.81e-03 Calcium sulfate 6.08e-05 0 0 0 2.30e-05 1.16e-06 Calcium sulfate 0 3.28e-03 0 0 3.28e-03 1.65e-04 Chlorine 0 4.03e-02 0 0 4.03e-02 2.04e-03 Chromium OXIde 0 7.63e-05 0 0 7.63e-05 2.83e-06 Chromium OXide 0 5.62e-05 0 0 8.19e+00 4.135% Colar, uargate (in ground) 4.49e-03 0 0 8.19e+00 4.135% Colar	Bauxite	6.31e-04	0	0	0	6.31e-04	3.18e-05
Bentonite (in ground) $2.48e-03$ 0 0 $2.48e-03$ $1.25e-04$ Bora $5.20e-07$ 0 0 $5.20e-07$ $2.62e-08$ Boric acid 0 $4.73e-03$ $2.39e-04$ $2.62e-08$ Boric acid 0 $1.27e-01$ 0 0 $1.27e-01$ Calcium Chloride 0 $1.27e-01$ 0 0 $1.27e-01$ Calcium Sulfate $6.08e-05$ 0 0 0 $6.48e-05$ Calcium sulfate (CaSO4, ore) $2.30e-05$ 0 0 0 $2.30e-05$ Calcium Sulfate (CaSO4, ore) $2.30e-05$ 0 0 0 $2.30e-05$ Chlorine 0 $4.03e-02$ 0 0 $4.03e-02$ $2.04e-03$ Chromium (V1) 0 $7.63e-05$ 0 0 $4.03e-02$ $2.04e-03$ Chromium Oxide 0 $5.62e-05$ 0 0 $5.62e-05$ $2.83e-06$ Clay (in ground) $4.49e-03$ 0 0 $8.19e+00$ 4.1355 Coal, average (in ground) $4.49e-03$ 0 0 $8.67e-04$ $4.37e-05$ Cyclohexane 0 $1.88e-04$ 0 0 $1.33e-03$ $6.72e-05$ Diomethyl Formamide 0 0 $1.33e-03$ 0 0 $1.33e-03$ $6.72e-05$ Diomethyl Formamide 0 0 0 0 $1.32e-04$ 0 0 $1.32e-04$ Dolomite $1.36e-05$ 0 0 0 $1.32e-04$ 0 0 $1.32e-04$ <	Bauxite (Al2O3, ore)	1.10e-03	4.47e-02	0	-1.14e-04	4.57e-02	2.31e-03
Borax 5.20e-07 0 0 5.20e-07 2.62e-08 Boric acid 0 4.73e-03 0 0 4.73e-03 2.39e-04 Calcium Chloride 0 1.27e-01 0 0 1.27e-01 6.42e-03 Calcium Sulfate 6.08e-05 0 0 0 9.54e-02 4.81e-03 Calcium Sulfate 6.08e-05 0 0 0 6.08e-05 3.07e-06 Calcium Sulfate CaS0-05 0 0 0 2.30e-05 1.16e-06 Chroim Oxide 0 3.28e-03 0 0 3.28e-03 1.65e-04 Chromium Oxide 0 7.63e-05 0 0 7.63e-05 2.83e-06 Chromium Oxide 0 0 0 8.67e-04 4.37e-05 2.83e-06 Calçi ground) 4.49e-03 0 0 8.67e-04 4.37e-05 Cyclohexane 0 1.88e-04 0 0 8.67e-04 4.37e-05 Cyclohexane 0	Bentonite (in ground)	2.48e-03	0	0	0	2.48e-03	1.25e-04
Boric acid 0 4.73e-03 0 0 4.73e-03 2.39e-04 Calcium Chloride 0 1.27e-01 0 0 1.27e-01 6.42e-03 Calcium Mydroxide 0 9.54e-02 0 0 9.54e-02 4.81e-03 Calcium sulfate 6.08e-05 0 0 0 2.30e-05 1.16e-06 Calcium sulfate 0 3.28e-03 0 0 3.28e-03 1.65e-04 Chlorine 0 4.03e-02 0 0 4.03e-02 2.04e-03 Chromium Oxide 0 7.63e-05 0 0 7.63e-05 3.85e-06 Chromium Oxide 0 5.62e-05 0 0 5.62e-05 2.83e-06 Chaq (in ground) 4.49e-03 0 0 8.19e+00 4.135% 0 4.37e-05 Cyclohexane 0 1.88e-04 0 0 8.67e-04 4.37e-05 0 4.37e-05 2.20e-04 0 0 1.38e-03 6.72e-04 4	Borax	5.20e-07	0	0	0	5.20e-07	2.62e-08
Calcium Chloride 0 1.27e-01 0 0 1.27e-01 6.42e-03 Calcium hydroxide 0 9.54e-02 0 0 9.54e-02 4.81e-03 Calcium sulfate 6.08e-05 0 0 0 6.08e-05 3.07e-06 Calcium sulfate (CaSO4, ore) 2.30e-05 0 0 2.30e-05 1.16e-06 Carium Oxide 0 3.28e-03 0 0 3.28e-03 1.65e-04 Chlorine 0 4.03e-02 0 0 4.03e-02 2.04e-03 Chromium ore 1.09e-07 0 0 0 7.63e-05 2.83e-06 Chromium ore 1.09e-07 0 0 0 5.62e-05 2.83e-06 Clay (in ground) 4.49e-03 0 0 8.19e+00 41.35% Coal, average (in ground) 0 0 0 1.38e-04 9.48e-05 Deoiling agent (unspecified) 1.33e-03 0 0 1.33e-03 6.72e-05 Diometly Formamide	Boric acid	0	4.73e-03	0	0	4.73e-03	2.39e-04
Calcium hydroxide 0 9.54e-02 0 0 9.54e-02 4.81e-03 Calcium sulfate 6.08e-05 0 0 0 6.08e-05 3.07e-06 Calcium sulfate (CaSO4, ore) 2.30e-05 0 0 0 2.30e-05 1.16e-06 Cerium Oxide 0 3.28e-03 0 0 3.28e-03 1.65e-04 Choronium OXide 0 7.63e-05 0 0 4.03e-02 2.04e-03 Chromium ore 1.09e-07 0 0 0 7.63e-05 3.85e-06 Chromium oxide 0 5.62e-05 0 0 5.51e-09 Chromium oxide 0 0 0 8.19e+00 41.35% Coal, average (in ground) 4.49e-03 0 0 8.67e-04 4.37e-05 Cyclohexane 0 1.88e-04 0 0 1.88e-04 9.48e-06 Deoiling agent (unspecified) 1.33e-03 0 0 0 1.33e-03 6.72e-05 Diomite (ingr	Calcium Chloride	0	1.27e-01	0	0	1.27e-01	6.42e-03
Calcium sulfate 6.08e-05 0 0 6.08e-05 3.07e-06 Calcium sulfate (CaSO4, ore) 2.30e-05 0 0 2.30e-05 1.16e-06 Cerium Oxide 0 3.28e-03 0 0 3.28e-03 1.66e-06 Chlorine 0 4.03e-02 0 0 4.03e-02 2.04e-03 Chromium (V1) 0 7.63e-05 0 0 1.09e-07 5.51e-09 Chromium ore 1.09e-07 0 0 0 1.09e-03 2.83e-06 Clay (in ground) 4.49e-03 0 0 8.19e+00 8.19e+00 41.35% Coal, average (in ground) 0 0 0 0 8.67e-04 4.37e-05 Cyclohexane 0 1.38e-04 0 0 1.33e-03 6.72e-05 2.20e-06 Diothyl Formamide 0 4.36e-05 0 0 1.33e-03 6.72e-05 Dolomite 1.33e-03 0 0 0 3.32e-04 1.17e-05	Calcium hydroxide	0	9.54e-02	0	0	9.54e-02	4.81e-03
Calcium sulfate (CaSO4, ore) 2.30e-05 0 0 2.30e-05 1.16e-06 Cerium Oxide 0 3.28e-03 0 0 3.28e-03 1.65e-04 Chlorine 0 4.03e-02 0 0 4.03e-02 2.04e-03 Chromium (VI) 0 7.63e-05 0 0 7.63e-05 3.85e-06 Chromium ore 1.09e-07 0 0 0 1.09e-07 5.1e-09 Chromium Oxide 0 5.62e-05 0 0 5.62e-05 2.83e-06 Clay (in ground) 4.49e-03 0 0 8.19e+00 8.19e+00 41.35% Coal, average (in ground) 0 0 0 3.06e-03 3.06e-03 1.55e-04 Copper (Cu, ore) 8.67e-04 0 0 0 8.67e-04 4.37e-05 Cyclohexane 0 1.88e-04 9.48e-06 9.48e-06 9.48e-06 2.02e-06 1.01e-05 2.02e-06 1.01e-05 2.02e-06 1.01e-05 2.02e-06 1.01e-05 <td>Calcium sulfate</td> <td>6.08e-05</td> <td>0</td> <td>0</td> <td>0</td> <td>6.08e-05</td> <td>3.07e-06</td>	Calcium sulfate	6.08e-05	0	0	0	6.08e-05	3.07e-06
Cerium Oxide 0 3.28e-03 0 0 3.28e-03 1.65e-04 Chlorine 0 4.03e-02 0 0 4.03e-02 2.04e-03 Chromium (VI) 0 7.63e-05 0 0 7.63e-05 3.85e-06 Chromium ore 1.09e-07 0 0 0 1.09e-07 5.51e-09 Chromium Oxide 0 5.62e-05 0 0 5.62e-05 2.83e-06 Clay (in ground) 4.49e-03 0 0 8.19e+00 8.19e+00 41.35% Coal, average (in ground) 0 0 0 3.06e-03 3.06e-03 1.55e-04 Cyclohexane 0 1.88e-04 0 0 8.67e-04 4.37e-05 Dooling agent (unspecified) 1.33e-03 0 0 0 1.38e-04 9.48e-06 Dooling agent (unspecified) 1.33e-03 0 0 0 2.24e-04 0 2.24e-04 1.7e-05 Dolomite (in ground) 1.60e-05 0 0	Calcium sulfate (CaSO4, ore)	2.30e-05	0	0	0	2.30e-05	1.16e-06
Chlorine 0 4.03e-02 0 0 4.03e-02 2.04e-03 Chromium (VI) 0 7.63e-05 0 0 7.63e-05 3.85e-06 Chromium ore 1.09e-07 0 0 0 1.09e-07 5.51e-09 Chromium Oxide 0 5.62e-05 0 0 5.62e-05 2.83e-06 Clay (in ground) 4.49e-03 0 0 8.19e+00 8.19e+00 4.135% Coal, average (in ground) 0 0 0 3.06e-03 3.06e-03 1.55e-04 Copper (Cu, ore) 8.67e-04 0 0 0 8.67e-04 9.48e-06 Deoiling agent (unspecified) 1.33e-03 0 0 1.33e-03 6.72e-05 Dinethyl Formamide 0 4.36e-05 0 0 4.36e-05 2.20e-06 Diottyl Sebacate 2.32e-04 0 0 0 1.43e-10 7.24e-12 Dolomite (in ground) 1.60e-05 0 0 1.60e-05 8.05e-07	Cerium Oxide	0	3.28e-03	0	0	3.28e-03	1.65e-04
Chromium (VI) 0 7.63e-05 0 0 7.63e-05 3.85e-06 Chromium ore 1.09e-07 0 0 0 1.09e-07 5.51e-09 Chromium Oxide 0 5.62e-05 0 0 5.62e-05 2.83e-06 Clay (in ground) 4.49e-03 0 0 8.19e+00 8.19e+00 41.35% Coal, average (in ground) 0 0 0 3.06e-03 3.06e-03 1.55e-04 Copper (Cu, ore) 8.67e-04 0 0 0 8.67e-04 4.37e-05 Cyclohexane 0 1.88e-04 0 0 1.33e-03 6.72e-05 Dinethyl Formamide 0 4.36e-05 0 0 4.36e-05 2.20e-06 Dioctyl Sebacate 2.32e-04 0 0 0 2.32e-04 1.17e-05 Dolomite 1.43e-10 0 0 0 1.43e-10 7.24e-12 Dolomite (in ground) 1.60e-05 0 0 0 3.12e-04 1.	Chlorine	0	4.03e-02	0	0	4.03e-02	2.04e-03
Chromium ore 1.09e-07 0 0 0 1.09e-07 5.51e-09 Chromium Oxide 0 5.62e-05 0 0 5.62e-05 2.83e-06 Clay (in ground) 4.49e-03 0 0 8.19e+00 8.19e+00 41.35% Coal, average (in ground) 0 0 0 3.06e-03 3.06e-03 1.55e-04 Copper (Cu, ore) 8.67e-04 0 0 8.67e-04 4.37e-05 Cyclohexane 0 1.88e-04 0 0 1.88e-04 9.48e-06 Deoiling agent (unspecified) 1.33e-03 0 0 1.33e-03 6.72e-05 Dimethyl Formamide 0 4.36e-05 0 0 4.36e-05 2.20e-06 Dioctyl Sebacate 2.32e-04 0 0 0 1.43e-10 7.24e-12 Dolomite 1.43e-10 0 0 0 1.60e-05 8.05e-07 Explosive (unspecified) 3.12e-04 0 0 0 3.12e-04 1.57e-05 <td>Chromium (VI)</td> <td>0</td> <td>7.63e-05</td> <td>0</td> <td>0</td> <td>7.63e-05</td> <td>3.85e-06</td>	Chromium (VI)	0	7.63e-05	0	0	7.63e-05	3.85e-06
Chromium Oxide 0 5.62e-05 0 0 5.62e-05 2.83e-06 Clay (in ground) 4.49e-03 0 0 8.19e+00 41.35% Coal, average (in ground) 0 0 0 3.06e-03 3.06e-03 1.55e-04 Copper (Cu, ore) 8.67e-04 0 0 0 8.67e-04 4.37e-05 Cyclohexane 0 1.88e-04 0 0 1.88e-04 9.48e-06 Deoiling agent (unspecified) 1.33e-03 0 0 0 1.33e-03 6.72e-05 Dimethyl Formamide 0 4.36e-05 0 0 4.36e-05 2.20e-06 Dioctyl Sebacate 2.32e-04 0 0 0 2.32e-04 1.17e-05 Dolomite 1.43e-10 0 0 0 1.43e-10 7.24e-12 Dolomite (in ground) 1.60e-05 0 0 0 9.96e-05 5.03e-06 Explosive (unspecified) 3.12e-04 0 0 0 3.75e-05 5.	Chromium ore	1.09e-07	0	0	0	1.09e-07	5.51e-09
Clay (in ground) $4.49e-03$ 0 $8.19e+00$ $8.19e+00$ 4.135% Coal, average (in ground)000 $3.06e-03$ $3.06e-03$ $1.55e-04$ Copper (Cu, ore) $8.67e-04$ 000 $8.67e-04$ $4.37e-05$ Cyclohexane0 $1.88e-04$ 00 $1.88e-04$ $9.48e-06$ Deoiling agent (unspecified) $1.33e-03$ 000 $1.33e-03$ $6.72e-05$ Dimethyl Formamide0 $4.36e-05$ 00 $4.36e-05$ $2.20e-06$ Dioctyl Sebacate $2.32e-04$ 000 $2.32e-04$ $1.17e-05$ Dolomite (in ground) $1.60e-05$ 000 $1.43e-10$ $7.24e-12$ Dolomite (in ground) $1.60e-05$ 00 0 $9.96e-05$ $8.05e-07$ Etxy Naphtol Sulphonic Acid (ENSA) $9.96e-05$ 00 0 $1.37e-01$ $6.93e-03$ Fluorocarbon resin0 $3.75e-05$ 00 $3.75e-05$ $1.89e-06$ Fluorocarbon resin0 $6.60e-03$ 0 0 $6.60e-03$ $3.33e-04$ Formaldehyde0 $6.60e-03$ 0 0 0 $1.69e-06$ $8.55e-08$ Formaldehyde0 $6.60e-03$ 0 0 0 $1.69e-06$ $8.55e-08$	Chromium Oxide	0	5.62e-05	0	0	5.62e-05	2.83e-06
Coal, average (in ground) 0 0 0 3.06e-03 3.06e-03 1.55e-04 Copper (Cu, ore) 8.67e-04 0 0 0 8.67e-04 4.37e-05 Cyclohexane 0 1.88e-04 0 0 1.88e-04 9.48e-06 Deoiling agent (unspecified) 1.33e-03 0 0 0 1.33e-03 6.72e-05 Dimethyl Formamide 0 4.36e-05 0 0 4.36e-05 2.20e-06 Dioctyl Sebacate 2.32e-04 0 0 0 2.32e-04 1.17e-05 Dolomite 1.43e-10 0 0 0 1.43e-10 7.24e-12 Dolomite (in ground) 1.60e-05 0 0 0 9.96e-05 5.03e-06 Explosive (unspecified) 3.12e-04 0 0 0 3.12e-04 1.57e-05 Ferric chloride 0 1.37e-01 0 0 3.75e-05 1.89e-06 Fluorocarbon resin 0 3.75e-05 0 0 3.75	Clay (in ground)	4.49e-03	0	0	8.19e+00	8.19e+00	41.35%
Copper (Cu, ore) 8.67e-04 0 0 8.67e-04 4.37e-05 Cyclohexane 0 1.88e-04 0 0 1.88e-04 9.48e-06 Deoiling agent (unspecified) 1.33e-03 0 0 0 1.33e-03 6.72e-05 Dimethyl Formamide 0 4.36e-05 0 0 4.36e-05 2.20e-06 Dioctyl Sebacate 2.32e-04 0 0 0 2.32e-04 1.17e-05 Dolomite (in ground) 1.60e-05 0 0 0 1.43e-10 7.24e-12 Dolomite (in ground) 1.60e-05 0 0 0 9.96e-05 5.03e-06 Explosive (unspecified) 3.12e-04 0 0 0 3.12e-04 1.57e-05 Ferric chloride 0 1.37e-01 0 0 3.75e-05 1.89e-06 Fluorocarbon resin 0 3.75e-05 0 0 3.33e-04 Fluorspar (CaF2, ore) 1.69e-06 0 0 6.60e-03 3.33e-04	Coal, average (in ground)	0	0	0	3.06e-03	3.06e-03	1.55e-04
Cyclohexane 0 1.88e-04 0 0 1.88e-04 9.48e-06 Deoiling agent (unspecified) 1.33e-03 0 0 0 1.33e-03 6.72e-05 Dimethyl Formamide 0 4.36e-05 0 0 4.36e-05 2.20e-06 Dioctyl Sebacate 2.32e-04 0 0 0 2.32e-04 1.17e-05 Dolomite 1.43e-10 0 0 0 1.43e-10 7.24e-12 Dolomite (in ground) 1.60e-05 0 0 0 1.60e-05 8.05e-07 Etxy Naphtol Sulphonic Acid (ENSA) 9.96e-05 0 0 9.96e-05 5.03e-06 Explosive (unspecified) 3.12e-04 0 0 0 3.12e-04 1.57e-05 Ferric chloride 0 1.37e-01 0 0 3.75e-05 1.89e-06 Fluorocarbon resin 0 3.75e-05 0 0 3.75e-05 1.89e-06 Formaldehyde 0 6.60e-03 0 0 6.60e-03	Copper (Cu, ore)	8.67e-04	0	0	0	8.67e-04	4.37e-05
Deoiling agent (unspecified) $1.33e-03$ 000 $1.33e-03$ $6.72e-05$ Dimethyl Formamide0 $4.36e-05$ 00 $4.36e-05$ $2.20e-06$ Dioctyl Sebacate $2.32e-04$ 000 $2.32e-04$ $1.17e-05$ Dolomite $1.43e-10$ 000 $1.43e-10$ $7.24e-12$ Dolomite (in ground) $1.60e-05$ 000 $1.60e-05$ $8.05e-07$ Etoxy Naphtol Sulphonic Acid (ENSA) $9.96e-05$ 000 $9.96e-05$ $5.03e-06$ Explosive (unspecified) $3.12e-04$ 000 $3.12e-04$ $1.57e-05$ Ferric chloride0 $1.37e-01$ 00 $3.75e-05$ $1.89e-06$ Fluorocarbon resin0 $3.75e-05$ 00 $3.75e-05$ $1.89e-06$ Formaldehyde0 $6.60e-03$ 0 0 $6.60e-03$ $3.33e-04$ Givcol ethers0 $2.35e-02$ 00 $2.35e-02$ $1.19e-03$	Cyclohexane	0	1.88e-04	0	0	1.88e-04	9.48e-06
Dimethyl Formamide 0 4.36e-05 0 0 4.36e-05 2.20e-06 Dioctyl Sebacate 2.32e-04 0 0 0 2.32e-04 1.17e-05 Dolomite 1.43e-10 0 0 0 1.43e-10 7.24e-12 Dolomite (in ground) 1.60e-05 0 0 0 1.60e-05 8.05e-07 Etoxy Naphtol Sulphonic Acid (ENSA) 9.96e-05 0 0 9.96e-05 5.03e-06 Explosive (unspecified) 3.12e-04 0 0 0 3.12e-04 1.57e-05 Ferric chloride 0 1.37e-01 0 0 3.75e-05 1.89e-06 Fluorocarbon resin 0 3.75e-05 0 0 1.69e-06 8.55e-08 Formaldehyde 0 6.60e-03 0 0 6.60e-03 3.33e-04 Glycol ethers 0 2.35e-02 0 0 2.35e-02 1.19e-03	Deoiling agent (unspecified)	1.33e-03	0	0	0	1.33e-03	6.72e-05
Dioctyl Sebacate 2.32e-04 0 0 2.32e-04 1.17e-05 Dolomite 1.43e-10 0 0 0 1.43e-10 7.24e-12 Dolomite (in ground) 1.60e-05 0 0 0 1.60e-05 8.05e-07 Etoxy Naphtol Sulphonic Acid (ENSA) 9.96e-05 0 0 0 9.96e-05 5.03e-06 Explosive (unspecified) 3.12e-04 0 0 0 3.12e-04 1.57e-05 Ferric chloride 0 1.37e-01 0 0 3.75e-05 1.89e-06 Fluorocarbon resin 0 3.75e-05 0 0 3.75e-06 8.55e-08 Formaldehyde 0 6.60e-03 0 0 6.60e-03 3.33e-04 Giveol ethers 0 2.35e-02 0 0 2.35e-02 1.19e-03	Dimethyl Formamide	0	4.36e-05	0	0	4.36e-05	2.20e-06
Dolomite 1.43e-10 0 0 1.43e-10 7.24e-12 Dolomite (in ground) 1.60e-05 0 0 0 1.60e-05 8.05e-07 Etoxy Naphtol Sulphonic Acid (ENSA) 9.96e-05 0 0 0 9.96e-05 5.03e-06 Explosive (unspecified) 3.12e-04 0 0 0 3.12e-04 1.57e-05 Ferric chloride 0 1.37e-01 0 0 1.37e-01 6.93e-03 Fluorocarbon resin 0 3.75e-05 0 0 3.75e-05 1.89e-06 Formaldehyde 0 6.60e-03 0 0 6.60e-03 3.33e-04 Giveol ethers 0 2.35e-02 0 0 2.35e-02 1.19e-03	Dioctyl Sebacate	2.32e-04	0	0	0	2.32e-04	1.17e-05
Dolomite (in ground) 1.60e-05 0 0 1.60e-05 8.05e-07 Etoxy Naphtol Sulphonic Acid (ENSA) 9.96e-05 0 0 0 9.96e-05 5.03e-06 Explosive (unspecified) 3.12e-04 0 0 0 3.12e-04 1.57e-05 Ferric chloride 0 1.37e-01 0 0 1.37e-01 6.93e-03 Fluorocarbon resin 0 3.75e-05 0 0 3.75e-05 1.89e-06 Fluorspar (CaF2, ore) 1.69e-06 0 0 0 6.60e-03 3.33e-04 Glycol ethers 0 2.35e-02 0 0 2.35e-02 1.19e-03	Dolomite	1.43e-10	0	0	0	1.43e-10	7.24e-12
Etoxy Naphtol Sulphonic Acid (ENSA) 9.96e-05 0 0 9.96e-05 5.03e-06 Explosive (unspecified) 3.12e-04 0 0 0 3.12e-04 1.57e-05 Ferric chloride 0 1.37e-01 0 0 1.37e-01 6.93e-03 Fluorocarbon resin 0 3.75e-05 0 0 3.75e-05 1.89e-06 Fluorspar (CaF2, ore) 1.69e-06 0 0 0 6.60e-03 3.33e-04 Glycol ethers 0 2.35e-02 0 0 2.35e-02 1.19e-03	Dolomite (in ground)	1.60e-05	0	0	0	1.60e-05	8.05e-07
Explosive (unspecified)3.12e-040003.12e-041.57e-05Ferric chloride01.37e-01001.37e-016.93e-03Fluorocarbon resin03.75e-05003.75e-051.89e-06Fluorspar (CaF2, ore)1.69e-060001.69e-068.55e-08Formaldehyde06.60e-03006.60e-033.33e-04Glycol ethers02.35e-02002.35e-021.19e-03	Etoxy Naphtol Sulphonic Acid (ENSA)	9.96e-05	0	0	0	9.96e-05	5.03e-06
Ferric chloride 0 1.37e-01 0 0 1.37e-01 6.93e-03 Fluorocarbon resin 0 3.75e-05 0 0 3.75e-05 1.89e-06 Fluorspar (CaF2, ore) 1.69e-06 0 0 0 1.69e-06 8.55e-08 Formaldehyde 0 6.60e-03 0 0 6.60e-03 3.33e-04 Glycol ethers 0 2.35e-02 0 0 2.35e-02 1.19e-03	Explosive (unspecified)	3.12e-04	0	0	0	3.12e-04	1.57e-05
Fluorocarbon resin 0 3.75e-05 0 0 3.75e-05 1.89e-06 Fluorspar (CaF2, ore) 1.69e-06 0 0 0 1.69e-06 8.55e-08 Formaldehyde 0 6.60e-03 0 0 6.60e-03 3.33e-04 Glycol ethers 0 2.35e-02 0 0 2.35e-02 1.19e-03	Ferric chloride	0	1.37e-01	0	0	1.37e-01	6.93e-03
Fluorspar (CaF2, ore) 1.69e-06 0 0 0 1.69e-06 8.55e-08 Formaldehyde 0 6.60e-03 0 0 6.60e-03 3.33e-04 Glycol ethers 0 2.35e-02 0 0 2.35e-02 1.19e-03	Fluorocarbon resin	0	3.75e-05	0	0	3.75e-05	1.89e-06
Formaldehyde 0 $6.60e-03$ 0 0 $6.60e-03$ $3.33e-04$ Glycol ethers 0 $2.35e-02$ 0 0 $2.35e-02$ $1.19e-03$	Fluorspar (CaF2, ore)	1.69e-06	0	0	0	1.69e-06	8.55e-08
Given ethers $0 = 235e_{-}02 = 0 = 0 = 235e_{-}02 = 119e_{-}03$	Formaldehyde	0	6.60e-03	0	0	6.60e-03	3.33e-04
	Glycol ethers	0	2.35e-02	0	0	2.35e-02	1.19e-03
Gravel/Sand 6.82e-03 0 0 0 6.82e-03 3.44e-04	Gravel/Sand	6.82e-03	0	0	0	6.82e-03	3.44e-04

Table J-2. CRT ancillary materials (kg/functional unit)

Material	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
HV Carbon (paste)	0	1.14e-05	0	0	1.14e-05	5.75e-07
Hydrochloric acid	0	2.36e-01	0	0	2.36e-01	1.19%
Hydrofluoric acid	0	8.65e-02	0	0	8.65e-02	4.36e-03
Hydrogen peroxide	0	8.45e-02	0	0	8.45e-02	4.26e-03
Iron (Fe, ore)	7.23e-02	0	0	3.41e-03	7.57e-02	3.82e-03
Iron ore	1.27e-04	0	0	0	1.27e-04	6.41e-06
Iron sulfate (FeSO4, ore)	2.56e-05	0	0	0	2.56e-05	1.29e-06
Isopentylacetate	0	1.74e-03	0	0	1.74e-03	8.80e-05
Isopropyl alcohol	0	1.94e-02	0	0	1.94e-02	9.80e-04
Lead (Pb, ore)	6.50e-05	0	0	0	6.50e-05	3.28e-06
Lime	0	3.04e-02	1.06e+00	1.06e-04	1.09e+00	5.49%
Limestone	0	6.91e-02	2.41e+00	2.41e-04	2.48e+00	12.51%
Limestone (CaCO3, in ground)	8.60e-01	1.08e+00	0	-2.39e-01	1.70e+00	8.58%
Lubricant (unspecified)	4.11e-02	0	0	0	4.11e-02	2.07e-03
Magnesium	5.08e-04	0	0	0	5.08e-04	2.56e-05
Maize	1.14e-04	0	0	0	1.14e-04	5.74e-06
Manganese (Mn, ore)	6.35e-08	0	0	0	6.35e-08	3.21e-09
Muratic Acid (drum)	0	1.87e-03	0	0	1.87e-03	9.41e-05
Natural gas (in ground)	0	0	0	4.52e-03	4.52e-03	2.28e-04
Nickel (Ni, ore)	2.45e-08	0	0	0	2.45e-08	1.24e-09
Nitric acid	0	1.44e-01	0	0	1.44e-01	7.26e-03
Nitrogen	0	4.57e-02	0	0	4.57e-02	2.31e-03
Oil (in ground)	0	0	0	3.35e-02	3.35e-02	1.69e-03
Olivine	1.08e-10	0	0	0	1.08e-10	5.43e-12
Olivine ore	1.24e-05	0	0	0	1.24e-05	6.27e-07
Oxalic acid	0	5.35e-05	0	0	5.35e-05	2.70e-06
Oxygen (Liquid)	0	7.57e-03	0	0	7.57e-03	3.82e-04
Periodic Acid	0	2.26e-04	0	0	2.26e-04	1.14e-05
Phenolsulphonic Acid	3.12e-03	0	0	0	3.12e-03	1.57e-04
Polyethylene glycol	0	5.04e-02	0	0	5.04e-02	2.55e-03
Polyvinyl alcohol	0	8.11e-03	0	0	8.11e-03	4.09e-04
Polyvinyl Pyrrolidone (PVP)	0	2.41e-02	0	0	2.41e-02	1.22e-03
Potassium chloride (KCl, as K2O, in ground)	1.92e-03	0	0	0	1.92e-03	9.69e-05
Potassium hydroxide	0	4.27e-02	0	0	4.27e-02	2.15e-03
Potassium permanganate	0	1.16e-03	0	0	1.16e-03	5.87e-05
Potassium peroxymonosulfate	0	7.06e-02	0	0	7.06e-02	3.56e-03
Potatoes	3.06e-05	0	0	0	3.06e-05	1.54e-06
Pumice	0	7.86e-02	0	0	7.86e-02	3.97e-03
PWB-solder mask solids	0	4.37e-02	0	0	4.37e-02	2.20e-03
Pyrite (FeS2, ore)	1.94e-01	0	0	0	1.94e-01	9.77e-03
Raw materials (unspecified)	8.16e-03	0	0	0	8.16e-03	4.12e-04
Sand (in ground)	5.85e-02	2.74e-02	0	2.71e+00	2.80e+00	14.13%
Silver (Ag, ore)	2.75e-09	0	0	0	2.75e-09	1.39e-10
Sodium Carbonate	0	3.22e-02	0	0	3.22e-02	1.63e-03
Sodium chloride	0	0	0	1.55e-04	1.55e-04	7.80e-06
Sodium chloride (NaCl, in ground or in sea)	7.61e-01	1.26e-02	0	-3.07e-05	7.73e-01	3.90%
Sodium Dichromate	2.63e-04	1.05e-04	0	0	3.68e-04	1.86e-05
Sodium Dichromate Dihydrate (VI)	0	3.10e-05	0	0	3.10e-05	1.56e-06

 Table J-2. CRT ancillary materials (kg/functional unit)

Material	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Sodium hydroxide	0	1.98e-01	0	0	1.98e-01	9.97e-03
Sodium Hypochlorite	0	9.25e-05	0	0	9.25e-05	4.67e-06
Sodium Metabisulfite	0	4.67e-03	0	0	4.67e-03	2.36e-04
Sodium Persulfate	0	3.54e-04	0	0	3.54e-04	1.79e-05
Sulfur	3.28e-03	0	0	0	3.28e-03	1.66e-04
Sulfur (S, in ground)	5.39e-03	0	0	0	5.39e-03	2.72e-04
Sulfuric acid	0	2.38e-01	0	0	2.38e-01	1.20%
Sulfuric acid, aluminum salt	0	6.75e-02	0	0	6.75e-02	3.41e-03
Surfactant, unspecified	0	1.42e-04	0	0	1.42e-04	7.17e-06
Synthetic resin, unspecified	0	8.53e-04	0	0	8.53e-04	4.30e-05
Talcum (ore)	8.89e-03	0	0	0	8.89e-03	4.49e-04
Tin (Sn, ore)	2.43e-02	0	0	0	2.43e-02	1.23e-03
Toluene	0	4.80e-03	0	0	4.80e-03	2.42e-04
unspecified CRT process material	0	5.77e-03	0	0	5.77e-03	2.91e-04
Uranium (U, ore)	0	0	0	4.16e-08	4.16e-08	2.10e-09
Wastepaper	1.70e-03	0	0	0	1.70e-03	8.59e-05
Xylene (mixed isomers)	0	4.80e-04	0	0	4.80e-04	2.42e-05
Zinc (Zn, ore)	3.79e-02	0	0	0	3.79e-02	1.91e-03
Total ancillary materials	2.11e+00	3.54e+00	3.47e+00	1.07e+01	1.98e+01	100.00%

 Table J-2. CRT ancillary materials (kg/functional unit)

APPENDIX J

Tuble 5 of ORT utility inputs										
Material	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total				
Fuels (kg/functional unit):										
Fuel Oil #2 (distillate)	0	1.16e+00	0	0	1.16e+00	0.27%				
Fuel oil #4 (avg. of #2 + #6)	0	1.37e-01	0	-1.38e+00	-1.24e+00	-0.29%				
Fuel oil #6 (residual)	0	3.68e+00	0	0	3.68e+00	0.85%				
LNG	0	3.35e-01	0	0	3.35e-01	0.08%				
LPG	0	3.51e+02	0	3.03e-03	3.51e+02	81.10%				
Natural gas	0	2.44e+00	0	-1.30e+00	1.14e+00	0.26%				
Coal, average (in ground)	2.25e+00	1.36e+01	0	-1.16e-02	1.58e+01	3.66%				
Coal, lignite (in ground)	9.73e-01	0	0	0	9.73e-01	0.22%				
Natural gas (in ground)	2.76e+00	4.56e+01	0	-2.09e-01	4.82e+01	11.14%				
Petroleum (in ground)	2.02e+00	9.71e+00	0	-5.77e-02	1.17e+01	2.70%				
Uranium (U, ore)	1.21e-04	2.29e-04	0	-1.99e-07	3.49e-04	8.06E-07				
Total fuel	s 8.00e+00	4.28e+02	0	-2.95e+00	4.33e+02	100.00%				
Electricity (MJ/functional unit):										
Electricity	7.32e+01	1.29e+02	2.29e+03	2.29e-01	2.49e+03					
Water (kg or L/functional unit):										
Water	5.54e+02	1.14e+04	1.14e+03	-2.73e+01	1.31e+04					
Total energy (fuels and electricity, MJ/funct	ional unit):									
Energy	3.66e+02	1.83e+04	2.29e+03	-1.28e+02	2.08e+04					

Table J-3. CRT utility inputs

Material	Upstream	Mfg Use		EOL	Total	% or Fraction of Total
1,1,1-Trichloroethane	0	2.31e-07	1.92e-06	-3.01e-08	2.12e-06	3.19e-09
1,2-Dichloroethane		3.74e-07	3.59e-06	-5.75e-08	3.90e-06	5.88e-09
1,2-Dichlorotetrafluoroethane	3.38e-07	0	0	0	3.38e-07	5.09e-10
1,4-Dichlorobenzene	0	3.06e-07	0	-2.26e-09	3.03e-07	4.57e-10
2,3,7,8-TCDD	0	3.73e-14	1.28e-12	1.28e-16	1.32e-12	1.99e-15
2,3,7,8-TCDF	0	1.31e-13	4.57e-12	4.58e-16	4.71e-12	7.09e-15
2,4-Dinitrotoluene	0	2.62e-09	2.51e-08	-4.22e-10	2.73e-08	4.11e-11
2-Chloroacetophenone	0	6.55e-08	6.28e-07	-1.46e-10	6.93e-07	1.04e-09
2-Methylnaphthalene	0	6.39e-09	2.67e-09	-5.37e-11	9.01e-09	1.36e-11
3-Methylcholanthrene	0	4.59e-10	0	-4.04e-12	4.54e-10	6.84e-13
5-Methyl chrysene	0	2.06e-10	1.97e-09	-3.32e-11	2.15e-09	3.23e-12
Acenaphthene	0	1.05e-08	5.68e-08	-7.77e-10	6.65e-08	1.00e-10
Acenaphthylene	0	2.86e-09	2.26e-08	-3.81e-10	2.50e-08	3.77e-11
Acetaldehyde	2.16e-06	5.33e-06	5.11e-05	-8.59e-07	5.78e-05	8.70e-08
Acetic acid	1.86e-05	0	0	0	1.86e-05	2.80e-08
Acetone	2.14e-06	0	0	0	2.14e-06	3.22e-09
Acetophenone	0	1.40e-07	1.35e-06	-2.26e-08	1.46e-06	2.20e-09
Acetylene	3.64e-05	0	0	0	3.64e-05	5.48e-08
Acrolein	2.38e-11	2.71e-06	2.60e-05	-4.37e-07	2.83e-05	4.26e-08
Adsorbable organic halides	1.14e-16	0	0	0	1.14e-16	1.71e-19
Alcohols	7.89e-06	0	0	0	7.89e-06	1.19e-08
Aldehydes	1.13e-04	1.52e-03	0	-1.63e-04	1.47e-03	2.21e-06
Alkane (unspecified)	3.37e-04	0	0	0	3.37e-04	5.07e-07
Alkenes	6.21e-05	0	0	0	6.21e-05	9.35e-08
Alkyne (unspecified)	4.10e-06	0	0	0	4.10e-06	6.18e-09
Aluminum (elemental)	3.35e-04	1.98e-05	0	-8.66e-08	3.55e-04	5.34e-07
Ammonia	4.36e-04	2.35e-03	0	-1.06e-04	2.68e-03	4.04e-06
Anthracene	0	2.88e-09	1.21e-10	-3.22e-10	2.68e-09	4.04e-12
Antimony	4.36e-07	1.66e-06	3.59e-06	-6.55e-08	5.61e-06	8.45e-09
Aromatic hydrocarbons	3.73e-04	5.29e-08	0	-1.37e-10	3.73e-04	5.61e-07
Arsenic	1.53e-04	1.54e-05	3.75e-05	-2.64e-06	2.03e-04	3.06e-07
Barium	5.18e-06	8.85e-07	2.06e-06	-2.45e-08	8.10e-06	1.22e-08
Benzaldehyde	1.26e-11	0	0	0	1.26e-11	1.90e-14
Benzene	6.17e-05	1.58e-02	1.17e-04	-7.31e-04	1.52e-02	2.29e-05
Benzo[a]anthracene	0	2.21e-09	9.27e-09	-1.25e-10	1.14e-08	1.71e-11
Benzo[a]pyrene	1.35e-06	7.92e-10	3.41e-09	-6.06e-11	1.35e-06	2.03e-09
Benzo[b,j,k]fluoranthene	0	1.30e-09	1.06e-08	-1.66e-10	1.18e-08	1.77e-11
Benzo[b]fluoranthene	0	5.07e-10	0	-4.23e-12	5.03e-10	7.58e-13
Benzo[g,h,i]perylene	0	1.05e-09	3.60e-09	-4.36e-11	4.61e-09	6.94e-12
Benzo[k]fluoranthene	0	5.07e-10	0	-4.23e-12	5.03e-10	7.58e-13
Benzyl chloride	0	6.55e-06	6.28e-05	-1.06e-06	6.83e-05	1.03e-07
Beryllium	7.37e-08	1.49e-06	1.91e-06	-3.06e-07	3.16e-06	4.76e-09
Biphenyl	0	1.59e-08	1.52e-07	-2.56e-09	1.66e-07	2.50e-10
Boron	7.51e-05	0	0	0	7.51e-05	1.13e-07
Bromine	3.80e-06	0	0	0	3.80e-06	5.72e-09
Bromium (Br)	7.09e-07	0	0	0	7.09e-07	1.07e-09
Bromoform	0	3.65e-07	3.50e-06	-5.88e-08	3.80e-06	5.73e-09
Bromomethane	0	1.50e-06	1.43e-05	-2.41e-07	1.56e-05	2.35e-08

Table J-4. CRT air pollutant emissions (kg/functional unit)

Material	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Butane	4.16e-05	5.35e-04	0	-4.72e-06	5.72e-04	8.61e-07
Butene	3.56e-07	0	0	0	3.56e-07	5.37e-10
Cadmium	6.60e-06	1.04e-06	4.82e-06	-6.37e-08	1.24e-05	1.87e-08
Calcium	3.01e-04	1.72e-05	0	-7.51e-08	3.18e-04	4.80e-07
Carbon	3.64e-07	0	0	0	3.64e-07	5.49e-10
Carbon dioxide	2.92e+01	1.79e+02	4.45e+02	2.59e+00	6.55e+02	98.68%
Carbon disulfide	0	1.22e-06	1.17e-05	-1.96e-07	1.27e-05	1.91e-08
Carbon monoxide	4.18e-02	4.58e-01	8.09e-02	-4.17e-03	5.76e-01	8.68e-04
Carbon tetrachloride	0	0	0	2.84e-09	2.84e-09	4.27e-12
CFC-13	5.04e-10	0	0	0	5.04e-10	7.58e-13
Chloride ions	3.06e-04	9.83e-05	1.82e-04	-2.54e-06	5.83e-04	8.78e-07
Chlorine	1.09e-06	5.84e-09	0	-2.72e-10	1.10e-06	1.65e-09
Chloroacetophenone	0	0	0	-1.04e-08	-1.04e-08	-1.57e-11
Chlorobenzene	0	2.06e-07	1.97e-06	-3.32e-08	2.15e-06	3.23e-09
Chloroform	0	5.52e-07	5.29e-06	-8.61e-08	5.76e-06	8.67e-09
Chromium	0	1.39e-07	0	0	1.39e-07	2.09e-10
Chromium (III)	7.72e-07	2.22e-05	2.45e-09	-2.14e-06	2.09e-05	3.14e-08
Chromium (VI)	7.72e-07	2.15e-05	7.21e-06	-2.14e-06	2.74e-05	4.12e-08
Chrysene	0	1.91e-09	1.02e-08	-1.55e-10	1.20e-08	1.80e-11
Cobalt	1.01e-06	3.92e-06	1.22e-05	-2.01e-07	1.69e-05	2.54e-08
Copper	4.71e-05	1.91e-06	9.96e-07	-2.54e-08	5.00e-05	7.52e-08
Cumene	0	4.35e-08	4.75e-07	-7.99e-09	5.11e-07	7.69e-10
Cumene hydroperoxide	0	6.06e-09	0	0	6.06e-09	9.12e-12
Cyanide (-1)	1.34e-07	2.34e-05	2.24e-04	-3.77e-06	2.44e-04	3.67e-07
Di(2-ethylhexyl)phthalate	0	6.83e-07	6.55e-06	-1.10e-07	7.12e-06	1.07e-08
Dibenzo[a,h]anthracene	0	6.71e-10	8.74e-10	-2.82e-12	1.54e-09	2.32e-12
Dichlorobenzene (mixed isomers)	0	0	0	-4.36e-10	-4.36e-10	-6.56e-13
Dichlorodifluoromethane	8.02e-10	0	0	0	8.02e-10	1.21e-12
Dichloromethane	0	2.71e-06	2.60e-05	-4.33e-07	2.83e-05	4.26e-08
Dimethyl Formamide	0	3.49e-05	0	0	3.49e-05	5.25e-08
Dimethyl sulfate	0	4.49e-07	4.30e-06	-7.24e-08	4.68e-06	7.05e-09
Dimethylbenzanthracene	0	3.82e-09	0	-3.39e-11	3.79e-09	5.70e-12
Dioxins, remaining unspeciated	2.79e-13	1.11e-10	5.84e-11	2.11e-10	3.81e-10	5.73e-13
Ethane	5.59e-04	7.90e-04		-6.97e-06	1.34e-03	2.02e-06
Ethanethiol	6.74e-07	0	0	0	6.74e-07	1.01e-09
Ethanol	9.49e-06	0	0	0	9.49e-06	1.43e-08
Ethyl Chloride	0	3.93e-07	3.77e-06	-6.33e-08	4.10e-06	6.17e-09
Ethylbenzene	2.25e-06	8.97e-07	8.46e-06	-1.32e-07	1.15e-05	1.73e-08
Ethylene	4.48e-04	0	0	0	4.48e-04	6.75e-07
Ethylene dibromide	0	1.12e-08	1.08e-07	-1.81e-09	1.17e-07	1.76e-10
Fluoranthene	0	8.56e-09	6.71e-08	-1.08e-09	7.46e-08	1.12e-10
Fluorene	0	1.02e-08	8.39e-08	-1.38e-09	9.28e-08	1.40e-10
Fluoride	2.12e-07	3.12e-07	1.95e-05	1.95e-09	2.01e-05	3.02e-08
Fluorides (F-)	6.70e-07	3.98e-05	0	-3.15e-07	4.01e-05	6.04e-08
Fluorine	8.94e-10	0	0	0	8.94e-10	1.35e-12
Formaldehyde	2.04e-05	1.24e-03	8.46e-05	-8.83e-06	1.33e-03	2.01e-06
Furans, remaining unspeciated	0	5.13e-10	9.32e-11	-1.14e-10	4.92e-10	7.41e-13
Halogenated hydrocarbons (unspecified)	2.27e-06	2.94e-13	0	-7.58e-16	2.27e-06	3.41e-09

Table J-4.	CRT air	pollutant	emissions	(kg/funct	ional unit)	
				(

Material	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Halogenated matter (unspecified)	1.86e-11	0	0	0	1.86e-11	2.79e-14
HALON-1301	5.81e-07	5.11e-10	0	-1.33e-12	5.82e-07	8.76e-10
HCFC-22	2.66e-09	0	0	0	2.66e-09	4.01e-12
Heptane	1.54e-05	0	0	0	1.54e-05	2.33e-08
Hexane	6.97e-06	4.59e-04	6.01e-06	-4.15e-06	4.68e-04	7.05e-07
HFC-125	4.54e-09	0	0	0	4.54e-09	6.84e-12
Hydrocarbons, remaining unspeciated	1.28e-02	1.58e-01	0	-6.12e-04	1.70e-01	2.57e-04
Hydrochloric acid	2.39e-03	1.12e-02	1.08e-01	-1.04e-03	1.20e-01	1.81e-04
Hydrofluoric acid	5.54e-04	1.40e-03	1.35e-02	-2.26e-04	1.52e-02	2.29e-05
Hydrogen	2.96e-04	0	0	0	2.96e-04	4.46e-07
Hydrogen cyanide	1.54e-10	0	0	0	1.54e-10	2.32e-13
Hydrogen sulfide	4.49e-05	3.11e-03	0	-1.31e-05	3.14e-03	4.73e-06
Indeno(1,2,3-cd)pyrene	0	1.50e-09	6.59e-09	-9.62e-11	7.99e-09	1.20e-11
Iodine	9.41e-07	0	0	0	9.41e-07	1.42e-09
Iron	2.36e-04	3.83e-05	0	-1.67e-07	2.74e-04	4.13e-07
Isophorone	0	5.43e-06	5.20e-05	-8.74e-07	5.66e-05	8.52e-08
Isopropylpropionate	4 01e-10	0	0	0	4 01e-10	6.04e-13
Lanthanum	8 40e-08	0	0	0	8 40e-08	1 2.7e-10
Lead	1.66e-03	1 30e-05	1 27e-05	1 42e-05	1 70e-03	2 55e-06
Lead (Pb. ore)	0	4 41e-07	0	0	4 41e-07	6.64e-10
Magnesium	1 57e-04	1.03e-04	9 86e-04	-1 66e-05	1.11e 07	1.85e-06
Manganese	1.97e 04	2 33e-05	4 56e-05	-4 99e-06	6 59e-05	9.93e-08
Manganese (Mn. ore)	0	1.09e-06	0	0	1.09e-06	1 64e-09
Mercantans	2 07e-07	0	0	0	2.07e-07	3 12e-10
Mercury	3.00e-06	1 12e-06	7 51e-06	-1 15e-07	1.15e-05	1.73e-08
Metals remaining unspeciated	1 21e-04	3 16e-07	7.510-00	-1.08e-09	1.13e-05	1.73e-08
Methane	6.40e.02	9.08e.01	6 459 01	-1.08C-09	1.220-04	2 37e 03
Methanol	6 73 - 06	9.080-01	0.450-01	-4.300-02	6 73e 06	1.01e.08
Methyl chloride	0.750-00	4 962 06	4 750 05	7 99e 07	5 17e 05	7.78e.08
Methyl ethyl ketone	0	4.90C-00	3 502 05	5 880 07	3 800 05	5 720 08
Methyl budrozine	0	1.502.06	1.520.05	-3.886-07	1.660.05	2.500.08
Methyl methoemilate	0	1.396-00	1.326-03	-2.306-07	1.000-05	2.506-08
Mathyl tart butyl athan	0	2.272.07	2 142 06	-5.020-08	2.412.06	2.946-09
Malada damana	4 (1- 07	3.276-07	5.146-00	-3.286-08	2.1500	3.146-09
Molyddenum	4.010-07	2.13e-00	1.822.06	-2.00e-08	2.422.06	4.756-09
Niekol	1.490-08	1 202 04	6.850.05	-2.920-08	2.420-00	2.6% 07
Nitragen digwide	4.13e-03	1.396-04	0.856-05	-4.396-00	5.050.02	5.00 0 -07
Nitrogen avides	5.700-02	6.05 - 01	1 182+00	1.00- 02	1.860+00	2.900-03
Nitrogen oxides	0.996-03	1.6(02	2 40- 02	-1.90e-02	1.800+00	2.808-03
Narrows oxide	9.476-04	1.00e-02	5.400-05	-1.11e-03	1.986-02	2.996-03
unspeciated	9.97e-02	1.10e-01	0	-1.91e-03	2.08e-01	3.13e-04
n-Propane	1.50e-04	1.69e-06	0	-3.70e-07	1.51e-04	2.27e-07
Other organics	5.60e-04	7.83e-02	0	-3.65e-03	7.52e-02	1.13e-04
o-xylene	0	1.13e-06	5.71e-08	-3.79e-08	1.15e-06	1.73e-09
Pentachlorobenzene	1.93e-15	0	0	0	1.93e-15	2.91e-18
Pentachlorophenol	3.12e-16	0	0	0	3.12e-16	4.71e-19
Pentane	3.47e-05	6.62e-04	0	-5.84e-06	6.91e-04	1.04e-06
Perfluoroethane	1.44e-05	0	0	0	1.44e-05	2.17e-08
Perfluoromethane	1.30e-04	0	0	0	1.30e-04	1.95e-07

Table J-4. CRT air pollutant emissions (kg/functional unit)

Material	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Phenanthrene	0	3.22e-08	2.51e-07	-4.11e-09	2.79e-07	4.20e-10
Phenol	2.93e-06	1.50e-07	1.43e-06	-2.41e-08	4.49e-06	6.77e-09
Phosphorus (yellow or white)	1.91e-06	1.26e-05	4.95e-06	-1.20e-07	1.94e-05	2.92e-08
Phosphorus pentoxide	8.18e-10	0	0	0	8.18e-10	1.23e-12
PM	1.28e-01	1.31e-01	0	-1.88e-02	2.40e-01	3.62e-04
PM-10	0	3.15e-03	5.78e-02	4.78e-06	6.09e-02	9.17e-05
Polycyclic aromatic hydrocarbons	1.87e-05	5.87e-11	0	-1.52e-13	1.87e-05	2.82e-08
Potassium	4.90e-05	0	0	0	4.90e-05	7.37e-08
Propionaldehyde	4.61e-13	3.55e-06	3.41e-05	-5.73e-07	3.71e-05	5.58e-08
Propionic acid	2.03e-10	0	0	0	2.03e-10	3.06e-13
Propylene	3.77e-05	0	0	0	3.77e-05	5.68e-08
Pyrene	0	5.44e-09	3.33e-08	-5.09e-10	3.82e-08	5.76e-11
Scandium	2.04e-08	0	0	0	2.04e-08	3.08e-11
Selenium	4.69e-07	1.29e-05	1.17e-04	-1.96e-06	1.28e-04	1.93e-07
Silicon	1.06e-03	1.72e-05	0	-7.51e-08	1.08e-03	1.62e-06
Sodium	2.88e-05	1.02e-04	0	-4.45e-07	1.30e-04	1.96e-07
Strontium	5.83e-06	0	0	0	5.83e-06	8.78e-09
Styrene	0	2.34e-07	2.24e-06	-3.77e-08	2.44e-06	3.67e-09
Sulfur dioxide	3.37e-01	1.26e-01	2.49e+00	8.30e-04	2.96e+00	4.45e-03
Sulfur oxides	5.71e-03	8.20e-01	0	-2.97e-02	7.96e-01	1.20e-03
Sulfuric acid	8.81e-07	0	0	0	8.81e-07	1.33e-09
Tars (unspecified)	9.87e-13	0	0	0	9.87e-13	1.49e-15
Tetrachloroethylene	0	4.02e-07	3.86e-06	-6.20e-08	4.20e-06	6.32e-09
Thallium	1.39e-08	0	0	0	1.39e-08	2.10e-11
Thorium	4.51e-08	0	0	0	4.51e-08	6.79e-11
Tin	2.62e-08	0	0	0	2.62e-08	3.94e-11
Titanium	1.00e-05	0	0	0	1.00e-05	1.51e-08
TOCs, remaining unspeciated	0	2.89e-04	5.76e-03	5.76e-07	6.05e-03	9.11e-06
Toluene	2.17e-05	3.84e-03	2.54e-05	-3.87e-07	3.89e-03	5.86e-06
Trichloroethylene	0	0	0	2.84e-09	2.84e-09	4.27e-12
Trichlorofluoromethane	8.27e-09	0	0	0	8.27e-09	1.25e-11
Uranium	4.43e-08	0	0	0	4.43e-08	6.67e-11
Vanadium	6.77e-05	2.74e-04	1.76e-05	-1.39e-06	3.57e-04	5.38e-07
Vinyl acetate	0	7.11e-08	6.82e-07	-1.15e-08	7.41e-07	1.12e-09
Vinyl chloride	0	0	0	5.67e-09	5.67e-09	8.54e-12
VOCs, remaining unspeciated	1.10e-03	0	0	0	1.10e-03	1.65e-06
Waste metals, unspecified	1.45e-05	0	0	0	1.45e-05	2.18e-08
Xylene (mixed isomers)	3.65e-05	3.84e-04	3.32e-06	-6.94e-09	4.24e-04	6.38e-07
Zinc (elemental)	5.84e-04	1.56e-05	1.52e-05	-2.78e-07	6.15e-04	9.26e-07
Zirconium	2.14e-08	0	0	0	2.14e-08	3.23e-11
Total air pollutants	3.00e+01	1.83e+02	4.49e+02	2.47e+00	6.64e+02	100.00%

Table J-4. CRT air pollutant emissions (kg/functional unit)

Material	Disposition	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
WASTEWATER STREAMS							
Wastewater stream	surface water	1.70e+01	1.41e+03	0	-3.65e+00	1.42e+03	93.76%
Wastewater stream	treatment	0	9.48e+01	0	0	9.48e+01	6.24%
Total wastewater streams	5	1.70e+01	1.51e+03	0.00e+00	-3.65e+00	1.52e+03	100.00%
WATER POLLUTANTS							
1,1,1-Trichloroethane	surface water	1.70e-13	0	0	0	1.70e-13	8.15e-15
1,2-Dichloroethane	surface water	0	0	0	8.41e-11	8.41e-11	4.03e-12
Acetic acid	surface water	1.49e-09	0	0	0	1.49e-09	7.16e-11
Acids (H+)	surface water	1.52e-04	1.03e-07	0	-4.78e-09	1.53e-04	7.30e-06
Adsorbable organic halides	surface water	1.04e-05	3.76e-10	0	-9.75e-13	1.04e-05	4.97e-07
Alcohols	surface water	1.19e-07	0	0	0	1.19e-07	5.71e-09
Aldehydes	surface water	3.10e-09	0	0	0	3.10e-09	1.48e-10
Alkane (unspecified)	surface water	6.77e-06	0	0	0	6.77e-06	3.24e-07
Alkenes	surface water	5.27e-07	0	0	0	5.27e-07	2.52e-08
Aluminum (+3)	surface water	2.74e-03	1.43e-04	0	-3.70e-07	2.89e-03	1.38e-04
Aluminum hydroxide	surface water	1.39e-09	0	0	0	1.39e-09	6.65e-11
Ammonia	surface water	1.86e-04	0	0	-4.05e-06	1.81e-04	8.69e-06
Ammonia ions	surface water	3.54e-06	2.76e-02	0	-6.63e-05	2.75e-02	1.32e-03
Aromatic hydrocarbons	surface water	3.00e-05	8.81e-08	0	-2.28e-10	3.01e-05	1.44e-06
Arsenic cmpds	surface water	5.47e-06	0	0	1.35e-08	5.48e-06	2.63e-07
Barium cmpds	surface water	3.53e-04	2.82e-07	0	4.03e-07	3.53e-04	1.69e-05
Barium sulfate	surface water	6.71e-04	0	0	0	6.71e-04	3.21e-05
Benzene	surface water	7.40e-06	0	0	8.41e-11	7.40e-06	3.54e-07
BOD	surface water	3.93e-04	1.95e-01	0	-4.65e-04	1.95e-01	9.34e-03
Boric acid	surface water	1.77e-06	0	0	0	1.77e-06	8.49e-08
Boron (B III)	surface water	1.75e-06	0	0	0	1.75e-06	8.39e-08
Cadmium cmpds	surface water	5.59e-07	2.94e-10	0	8.03e-10	5.60e-07	2.68e-08
Calcium (+2)	surface water	4.96e-03	0	0	0	4.96e-03	2.38e-04
Carbon tetrachloride	surface water	0	0	0	8.41e-11	8.41e-11	4.03e-12
Carbonate ion	surface water	4.83e-03	0	0	0	4.83e-03	2.31e-04
Cesium (+2)	surface water	1.93e-08	0	0	0	1.93e-08	9.24e-10
Chloride ions	surface water	4.29e-01	6.48e+00	0	-2.39e-02	6.88e+00	32.95%
Chlorine	surface water	8.94e-07	0	0	0	8.94e-07	4.28e-08
Chloroform	surface water	1.75e-11	0	0	8.41e-11	1.02e-10	4.87e-12
Chromium	surface water	8.20e-08	0	0	0	8.20e-08	3.93e-09
Chromium (III)	surface water	2.73e-05	3.77e-08	0	1.32e-08	2.74e-05	1.31e-06
Chromium (VI)	surface water	4.68e-07	3.77e-08	0	1.32e-08	5.19e-07	2.49e-08
Chromium ore	surface water	0	1.02e-05	0	0	1.02e-05	4.90e-07
Chromium ore	treatment	0	1.03e-06	0	0	1.03e-06	4.95e-08
Cobalt (Co I, Co II, Co III)	surface water	2.45e-06	0	0	0	2.45e-06	1.17e-07
COD	surface water	1.00e-02	1.60e+00	0	-3.94e-03	1.61e+00	7.71%
COD	treatment	0	8.33e-03	0	0	8.33e-03	3.99e-04
Copper	surface water	0	1.80e-06	0	0	1.80e-06	8.63e-08
Copper (+1 & +2)	surface water	1.61e-05	5.87e-09	0	-1.52e-11	1.61e-05	7.70e-07
Copper (+1 & +2)	treatment	0	9.71e-05	0	0	9.71e-05	4.65e-06
Cyanide (-1)	surface water	1.85e-06	6.06e-07	0	-1.07e-12	2.46e-06	1.18e-07
Dichloromethane	surface water	5.02e-08	0	0	1.35e-10	5.03e-08	2.41e-09
Dissolved organic matter (chlorinated)	surface water	2.27e-06	0	0	0	2.27e-06	1.08e-07
Dissolved organics	surface water	5.54e-04	2.71e-07	0	-1.26e-08	5.55e-04	2.66e-05

Table J-5.	CRT wat	er outputs	(kg/functiona	l unit)
------------	---------	------------	---------------	---------

	able J-3.		outputs (kg/func	uonai ui	iiii)	
Material	Disposition	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Dissolved solids	surface water	5.30e-03	3.62e+00	0	-8.77e-05	3.62e+00	17.36%
Dissolved solids	treatment	0	8.01e-02	0	0	8.01e-02	3.84e-03
Edetic acid (EDTA)	surface water	3.01e-09	0	0	0	3.01e-09	1.44e-10
Ethylbenzene	surface water	6.70e-07	0	0	3.04e-10	6.71e-07	3.21e-08
Fluoride	surface water	1.89e-05	3.45e-03	0	0	3.47e-03	1.66e-04
Fluoride	treatment	0	3.51e-04	0	0	3.51e-04	1.68e-05
Fluorides (F-)	surface water	9.63e-05	2.97e-03	0	-7.72e-06	3.06e-03	1.46e-04
Formaldehyde	surface water	2.20e-13	0	0	0	2.20e-13	1.05e-14
Halogenated matter (organic)	surface water	2.71e-10	1.17e-10	0	-3.05e-13	3.88e-10	1.86e-11
Hexachloroethane	surface water	3.09e-17	0	0	0	3.09e-17	1.48e-18
Hydrazine	surface water	1.38e-09	0	0	0	1.38e-09	6.62e-11
Hydrocarbons, remaining unspeciated	surface water	1.58e-04	3.52e-04	0	-4.25e-07	5.09e-04	2.44e-05
Hypochlorite	surface water	5.26e-09	0	0	0	5.26e-09	2.52e-10
Hypochlorous acid	surface water	5.26e-09	0	0	0	5.26e-09	2.52e-10
Iodide (-1)	surface water	3.23e-06	0	0	0	3.23e-06	1.54e-07
Iron	surface water	0	2.93e-03	0	0	2.93e-03	1.40e-04
Iron (+2 & +3)	surface water	1.76e-03	3.55e-07	0	-1.57e-08	1.76e-03	8.42e-05
Lead	surface water	0	4.64e-05	0	0	4.64e-05	2.22e-06
Lead	treatment	0	1.03e-06	0	0	1.03e-06	4.95e-08
Lead cmpds	surface water	1.59e-05	1.17e-09	0	1.60e-09	1.59e-05	7.59e-07
Lead cmpds	treatment	0	1.62e-05	0	0	1.62e-05	7.75e-07
Lithium salts	surface water	1.55e-10	0	0	0	1.55e-10	7.40e-12
Magnesium (+2)	surface water	2.68e-03	0	0	0	2.68e-03	1.28e-04
Manganese	surface water	0	3.60e-06	0	0	3.60e-06	1.73e-07
Manganese cmpds	surface water	1.02e-04	0	0	0	1.02e-04	4.88e-06
Mercury compounds	surface water	9.68e-07	1.35e-12	0	4.33e-11	9.68e-07	4.63e-08
Metals, remaining unspeciated	surface water	6.74e-04	9.75e-03	0	-3.42e-05	1.04e-02	4.98e-04
Molybdenum	surface water	0	1.20e-07	0	0	1.20e-07	5.77e-09
Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI)	surface water	8.27e-06	0	0	0	8.27e-06	3.96e-07
Morpholine	surface water	1.46e-08	0	0	0	1.46e-08	7.01e-10
Nickel	surface water	0	7.94e-05	0	0	7.94e-05	3.80e-06
Nickel (+2)	surface water	6.69e-07	0	0	0	6.69e-07	3.20e-08
Nickel cmpds	surface water	1.75e-05	5.87e-10	0	-1.52e-12	1.75e-05	8.38e-07
Nitrate	surface water	9.24e-05	6.10e-05	0	-1.96e-06	1.51e-04	7.25e-06
Nitrates/nitrites	surface water	2.29e-05	3.95e-06	0	0	2.69e-05	1.29e-06
Nitrites	surface water	1.91e-06	0	0	0	1.91e-06	9.15e-08
Nitrogen	surface water	4.46e-05	7.18e-03	0	0	7.23e-03	3.46e-04
Nitrogen dioxide	surface water	6.20e-04	0	0	0	6.20e-04	2.97e-05
Oil & grease	surface water	0	7.46e-03	0	0	7.46e-03	3.57e-04
Other nitrogen	surface water	5.28e-04	1.59e-08	0	-4.11e-11	5.28e-04	2.53e-05
Other organics	surface water	3.98e-05	0	0	0	3.98e-05	1.90e-06
Oxalic acid	surface water	6.02e-09	0	0	0	6.02e-09	2.88e-10
o-xvlene	surface water	0	0	0	7.84e-10	7.84e-10	3.76e-11
Phenol	surface water	6.25e-05	3.63e-03	0	-9.41e-06	3.68e-03	1.76e-04
Phosphate (PO43-)	surface water	1.75e-04	0	0	0	1.75e-04	8.39e-06
Phosphate as P2O5	surface water	0	1.21e-06	0	0	1.21e-06	5.80e-08
Phosphates	surface water	3.36e-05	5.85e-07	0	6.93e-09	3.42e-05	1.64e-06
Phosphorus (yellow or white)	surface water	3.94e-06	5.05e-05	0	0	5.44e-05	2.61e-06

Table J-5. CRT water outputs (kg/functional unit)

Material	Disposition	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Phosphorus pentoxide	surface water	2.45e-08	0	0	0	2.45e-08	1.18e-09
Polycyclic aromatic hydrocarbons	surface water	7.88e-06	1.41e-09	0	-3.66e-12	7.88e-06	3.77e-07
Potassium (+1)	surface water	1.12e-03	0	0	0	1.12e-03	5.38e-05
Rubidium ion (Rb+)	surface water	1.93e-07	0	0	0	1.93e-07	9.26e-09
Salts (unspecified)	surface water	1.71e-03	1.62e-03	0	-4.21e-06	3.33e-03	1.59e-04
Sand	surface water	1.80e-08	0	0	0	1.80e-08	8.62e-10
Saponifiable oils and fats	surface water	2.38e-04	0	0	0	2.38e-04	1.14e-05
Selenium	surface water	1.34e-05	0	0	8.15e-11	1.34e-05	6.44e-07
Silver compounds	surface water	1.16e-08	0	0	2.55e-09	1.41e-08	6.78e-10
Sodium	surface water	9.07e-05	0	0	0	9.07e-05	4.34e-06
Sodium (+1)	surface water	2.90e-01	7.04e+00	0	-3.08e-02	7.30e+00	34.94%
Strontium (Sr II)	surface water	3.73e-04	0	0	0	3.73e-04	1.79e-05
Sulfate ion (-4)	surface water	3.75e-02	9.61e-04	0	-1.85e-06	3.84e-02	1.84e-03
Sulfate ion (-4)	treatment	0	1.09e-03	6.84e-02	6.84e-06	6.95e-02	3.33e-03
Sulfide	surface water	3.99e-06	5.97e-08	0	-2.64e-09	4.05e-06	1.94e-07
Sulfites	surface water	1.73e-06	0	0	0	1.73e-06	8.27e-08
Sulfur	surface water	3.32e-11	0	0	0	3.32e-11	1.59e-12
Suspended solids	surface water	7.72e-03	8.69e-01	0	-2.11e-03	8.74e-01	4.19%
Suspended solids	treatment	0	1.29e-03	1.78e-03	1.78e-07	3.07e-03	1.47e-04
Tars (unspecified)	surface water	1.53e-14	0	0	0	1.53e-14	7.34e-16
Tetrachloroethylene	surface water	7.61e-14	0	0	8.41e-11	8.42e-11	4.03e-12
Tin (Sn++, Sn4+)	surface water	1.35e-04	0	0	0	1.35e-04	6.45e-06
Titanium tetrachloride	surface water	1.59e-04	0	0	0	1.59e-04	7.60e-06
TOCs	surface water	2.85e-03	8.81e-07	0	-2.28e-09	2.86e-03	1.37e-04
Toluene	surface water	8.52e-06	1.29e-08	0	2.90e-09	8.54e-06	4.09e-07
Trichloroethylene	surface water	4.68e-12	0	0	8.41e-11	8.88e-11	4.25e-12
Triethylene glycol	surface water	1.00e-05	0	0	0	1.00e-05	4.81e-07
Vanadium (V3+, V5+)	surface water	1.42e-05	0	0	0	1.42e-05	6.78e-07
Vinyl chloride	surface water	0	0	0	1.69e-10	1.69e-10	8.07e-12
VOCs, remaining unspeciated	surface water	6.75e-06	0	0	0	6.75e-06	3.23e-07
Waste metals, unspecified	surface water	2.14e-04	0	0	0	2.14e-04	1.03e-05
Waste oil	surface water	3.65e-03	1.01e-01	0	-3.13e-04	1.04e-01	4.98e-03
Xylene (mixed isomers)	surface water	2.17e-05	0	0	7.31e-10	2.17e-05	1.04e-06
Zinc (+2)	surface water	6.66e-05	1.23e-08	0	-3.17e-10	6.66e-05	3.19e-06
Zinc (elemental)	surface water	0	1.39e-05	0	0	1.39e-05	6.64e-07
Zinc (elemental)	treatment	0	1.03e-06	0	0	1.03e-06	4.95e-08
Total water pollutant	ts	8.12e-01	2.01e+01	7.02e-02	-6.18e-02	2.09e+01	100.00%

Table J-5. CRT water outputs (kg/functional unit)

Material	Disposition	Upstream	Mfg	Use	EOL	Total	% or Fraction
Barium debris (D008 waste)	landfill	0	2.14E-04	0	0	2.14E-04	2.26E-05
Broken CRT glass	landfill	0	1.88E-03	0	0	1.88E-03	1.99E-04
Carbon and filmer waste	landfill	0	1.03E-03	0	0	1.03E-03	1.09E-04
Chrome debris (D007 waste)	treatment	0	1.47E-04	0	0	1.47E-04	1.56E-05
Chrome liquid waste (D007 waste)	recycling/reuse	0	9.80E-03	0	0	9.80E-03	1.04E-03
cinders from CRT glass mfg (70% PbO)	landfill	0	8.26E-03	0	0	8.26E-03	8.73E-04
CRT glass faceplate EP dust (Pb) (D008 waste)	landfill	0	1.03E-03	0	0	1.03E-03	1.09E-04
CRT glass funnel EP dust (Pb) (D008 waste)	recycling/reuse	0	5.01E-03	0	0	5.01E-03	5.30E-04
CRT glass, cullet	recycling/reuse	0	0	0	4.84E-01	4.84E-01	5.12%
CRT glass, funnel	recycling/reuse	0	0	0	2.29E-01	2.29E-01	2.42%
EOL CRT Monitor, landfilled	landfill	0	0	0	7.20E+00	7.20E+00	76.11%
Frit	landfill	0	2.99E-03	0	0	2.99E-03	3.16E-04
General Hazardous Waste	landfill	4.85E-02	0	0	-9.61E-05	4.84E-02	5.12E-03
General Hazardous Waste	treatment	0	1.24E-01	0	0	1.24E-01	1.31E-02
Hazardous sludge (Pb) (D008)	landfill	0	1.52E-03	0	0	1.52E-03	1.60E-04
Hazardous waste	landfill	3.85E-04	6.15E-01	0	-1.50E-03	6.14E-01	6.49%
Hydrofluoric acid	landfill	0	1.78E-03	0	0	1.78E-03	1.88E-04
Lead contaminated grit (D008 waste)	landfill	0	3.46E-05	0	0	3.46E-05	3.65E-06
Lead debris (D008 waste)	landfill	0	2.14E-04	0	0	2.14E-04	2.26E-05
Lead sulfate cake	landfill	0	2.67E-05	0	0	2.67E-05	2.82E-06
Printed wiring board (PWB)	recycling/reuse	0	0	0	1.46E-01	1.46E-01	1.54E-02
PWB-Decontaminating debris	treatment	0	1.55E-02	0	0	1.55E-02	1.64E-03
PWB-Lead contaminated waste oil	treatment	0	1.16E-02	0	0	1.16E-02	1.23E-03
PWB-Route dust	recycling/reuse	0	1.20E-02	0	0	1.20E-02	1.27E-03
PWB-Solder dross	recycling/reuse	0	6.70E-02	0	0	6.70E-02	7.09E-03
PWB-Waste cupric etchant	recycling/reuse	0	2.25E-01	0	0	2.25E-01	2.38%
Silica coat waste	treatment	0	2.86E-04	0	0	2.86E-04	3.02E-05
Slag and ash	landfill	0	2.47E-03	0	0	2.47E-03	2.61E-04
sludge from CRT glass mfg (1% PbO)	landfill	0	8.77E-04	0	0	8.77E-04	9.28E-05
Slurry scrap (chromium-based)	landfill	0	8.62E-04	0	0	8.62E-04	9.12E-05
Spent solvent, unspecified	treatment	0	2.75E-04	0	0	2.75E-04	2.90E-05
Transformer	recycling/reuse	0	0	0	2.28E-01	2.28E-01	2.41%
Unspecified sludge	landfill	0	5.22E-03	0	0	5.22E-03	5.52E-04
Unspecified sludge	recycling/reuse	0	5.56E-03	0	0	5.56E-03	5.88E-04
Waste acid (mostly 3% HCl solution)	recycling/reuse	0	3.93E-03	0	0	3.93E-03	4.16E-04
Waste Batch (Ba, Pb) (D008 waste)	landfill	0	1.41E-03	0	0	1.41E-03	1.50E-04
Waste finishing sludge (Pb) (D008 waste)	landfill	0	2.56E-04	0	0	2.56E-04	2.70E-05
Waste oxygenated solvents	treatment	0	9.48E-05	0	0	9.48E-05	1.00E-05
Waste water treatment (WWT) filters	landfill	0	3.40E-04	0	0	3.40E-04	3.60E-05
Total hazardous wast	e	4.89E-02	1.13E+00	0.00E+00	8.28E+00	9.46E+00	100.00%

Table J-6. CRT hazardous waste outputs (kg/functional unit)

Material	Disposition	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
abrasive sludge	recycling/reuse	0	4.21E-02	0	0	4.21E-02	2.44E-04
acid absorbent	landfill	0	8.13E-05	0	0	8.13E-05	4.72E-07
Aluminum (elemental)	landfill	5.68E-05	0	0	0	5.68E-05	3.29E-07
Aluminum scrap	recycling/reuse	0	1.82E-04	0	1.10E-01	1.10E-01	6.37E-04
Aluminum scrap, Wabash 319	recycling/reuse	0	5.08E-07	0	0	5.08E-07	2.95E-09
Arsenic	landfill	2.27E-08	0	0	0	2.27E-08	1.32E-10
Bauxite residues	landfill	0	1.21E-02	0	-2.96E-05	1.21E-02	7.03E-05
blasting media	landfill	0	3.66E-04	0	0	3.66E-04	2.12E-06
Broken CRT glass	recycling/reuse	0	1.08E+00	0	0	1.08E+00	0.62%
Cables/wires	recycling/reuse	0	8.86E-03	0	0	8.86E-03	5.14E-05
Cadmium	landfill	1.65E-09	0	0	0	1.65E-09	9.59E-12
Calcium	landfill	2.27E-04	0	0	0	2.27E-04	1.32E-06
Carbon	landfill	1.75E-04	0	0	0	1.75E-04	1.01E-06
Carbon Steel Scrap	recycling/reuse	0	0	0	4.10E-01	4.10E-01	0.24%
Cardboard	recycling/reuse	9.81E-05	0	0	0	9.81E-05	5.69E-07
Chromium (III)	landfill	1.42E-07	0	0	0	1.42E-07	8.23E-10
Chromium (VI)	landfill	1.42E-07	0	0	0	1.42E-07	8.23E-10
Coal waste	landfill	0	1.46E+00	5.09E+01	5.09E-03	5.23E+01	30.37%
Cobalt	landfill	1.00E-09	0	0	0	1.00E-09	5.81E-12
Cobalt nitrate	treatment	0	6.10E-05	0	0	6.10E-05	3.54E-07
Copper	landfill	5.01E-09	0	0	0	5.01E-09	2.91E-11
Copper scrap	recycling/reuse	0	0	0	6.67E-02	6.67E-02	3.87E-04
CRT glass, faceplate	landfill	0	2.43E-02	0	0	2.43E-02	1.41E-04
CRT glass, faceplate	recycling/reuse	0	0	0	3.54E-01	3.54E-01	0.21%
Diesel fuel	treatment	0	4.07E-05	0	0	4.07E-05	2.36E-07
Dust	treatment	0	3.43E-03	0	0	3.43E-03	1.99E-05
Dust/sludge	landfill	0	5.64E-01	1.97E+01	1.97E-03	2.02E+01	11.75%
EOL CRT Monitor, incinerated	treatment	0	0	0	3.31E+00	3.31E+00	1.92%
EOL CRT Monitor, landfilled	landfill	0	0	0	3.91E+00	3.91E+00	2.27%
EOL CRT Monitor, recycled	recycling/reuse	0	0	0	2.42E+00	2.42E+00	1.40%
EOL CRT Monitor, remanufactured	recycling/reuse	0	0	0	6.60E-01	6.60E-01	0.38%
Ferric chloride	recycling/reuse	0	3.69E-01	0	0	3.69E-01	0.21%
FGD sludge	landfill	0	2.14E-01	0	-9.24E-04	2.14E-01	0.12%
flame retardant high-impact polystyrene (HIPS)	recycling/reuse	0	0	0	4.03E-01	4.03E-01	0.23%
Fly/bottom ash	landfill	0	3.65E-01	1.27E+01	1.27E-03	1.31E+01	7.59%
Iron	landfill	1.14E-04	0	0	0	1.14E-04	6.59E-07
Iron scrap	recycling/reuse	3.43E-01	0	0	2.50E+00	2.85E+00	1.65%
Lead	landfill	2.29E-08	0	0	0	2.29E-08	1.33E-10
Manganese	landfill	2.27E-06	0	0	0	2.27E-06	1.32E-08
Mercury	landfill	1.42E-10	0	0	0	1.42E-10	8.21E-13
Mineral waste	landfill	4.42E-01	2.61E-03	0	-6.76E-06	4.44E-01	0.26%
Mining waste	landfill	4.48E-01	0	0	-1.90E-06	4.48E-01	0.26%
Mixed industrial (waste)	landfill	4.87E-02	1.00E+00	0	-5.12E-04	1.05E+00	0.61%
Nickel	landfill	7.50E-09	0	0	0	7.50E-09	4.35E-11
Nickel nitrate	treatment	0	6.10E-05	0	0	6.10E-05	3.54E-07
Nitrogen	landfill	4.23E-08	0	0	0	4.23E-08	2.45E-10
Non mineral waste (inert)	landfill	2.95E-05	0	0	0	2.95E-05	1.71E-07
Non toxic chemical waste (unspecified)	landfill	4.07E-03	6.11E-04	0	-1.58E-06	4.68E-03	2.71E-05

Table J-7.	CRT solid waste	outputs (kg	<u>/functional unit)</u>

Material	Disposition	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Oily rags & filter media	landfill	0	3.25E-04	0	0	3.25E-04	1.89E-06
Oily rags & filter media	recycling/reuse	0	4.07E-05	0	0	4.07E-05	2.36E-07
parts cleaner solvent	recycling/reuse	0	8.13E-05	0	0	8.13E-05	4.72E-07
Phosphorus (yellow or white)	landfill	2.90E-06	0	0	0	2.90E-06	1.68E-08
Plating process sludge	landfill	0	3.28E-04	0	0	3.28E-04	1.90E-06
PM	landfill	0	5.33E-04	0	0	5.33E-04	3.09E-06
Potassium Carbonate	landfill	0	3.30E-03	0	0	3.30E-03	1.91E-05
Printed wiring board (PWB)	recycling/reuse	0	3.70E-02	0	0	3.70E-02	2.15E-04
PWB-Drill dust	landfill	0	1.49E-02	0	0	1.49E-02	8.67E-05
Sewage sludge (unspecified)	landfill	6.64E-05	0	0	0	6.64E-05	3.85E-07
Slag and ash	landfill	9.65E-02	6.66E+01	0	-1.49E+01	5.18E+01	30.06%
Slag and ash	recycling/reuse	0	6.85E-01	0	-3.01E-03	6.82E-01	0.40%
Sludge (aquadag)	landfill	0	2.22E-03	0	0	2.22E-03	1.29E-05
sludge (calcium fluoride, CaF2)	recycling/reuse	0	1.75E-02	0	0	1.75E-02	1.02E-04
Sludge (phosphor)	landfill	0	4.31E-03	0	0	4.31E-03	2.50E-05
Sodium Carbonate	landfill	0	3.29E-03	0	0	3.29E-03	1.91E-05
Spent solvents (toluene,xylene,dimethyl formamide,isopropyl alcohol)	recycling/reuse	0	4.17E-02	0	0	4.17E-02	2.42E-04
Stannous sludge	recycling/reuse	6.64E-04	0	0	0	6.64E-04	3.85E-06
Steel scrap (tinplated)	recycling/reuse	1.58E-01	0	0	0	1.58E-01	9.15E-04
Sulfur	landfill	2.55E-05	0	0	0	2.55E-05	1.48E-07
Unspecified sludge	landfill	0	7.69E-03	0	0	7.69E-03	4.46E-05
Unspecified sludge	recycling/reuse	0	1.26E-01	0	0	1.26E-01	7.29E-04
Unspecified solid waste	landfill	4.94E+00	0	0	-7.86E-01	4.15E+00	2.41%
Unspecified solid waste	recycling/reuse	3.07E+00	4.33E-01	0	0	3.50E+00	2.03%
Unspecified solid waste	treatment	0	3.66E+00	0	0	3.66E+00	2.12%
Unspecified solid waste (incinerated)	treatment	4.24E-03	1.33E-02	0	-5.82E-05	1.75E-02	1.01E-04
Unspecified waste	landfill	0	3.38E+00	0	-1.49E-02	3.36E+00	1.95%
Unspecified waste	recycling/reuse	0	0	0	-8.90E-02	-8.90E-02	-5.17E-04
Waste alkali (cleaning caustic and alkali soda	recycling/reuse	0	2.12E-02	0	0	2.12E-02	1.23E-04
effluent)		0	4.015.05	0	0	4.215.05	2.445.07
Waste alkali, unspecified	treatment	0	4.21E-05	0	0	4.21E-05	2.44E-07
Waste metals, unspecified	recycling/reuse	0	8.79E-02	0	0	8.79E-02	5.10E-04
Waste oil	landfill	8.15E-05	0	0	0	8.15E-05	4./3E-0/
Waste oil	recycling/reuse	0	1.43E-03	0	0	1.43E-03	8.29E-06
Waste oil	treatment	0	9.09E-03	0	0	9.09E-03	5.28E-05
Waste Plastic (packing material)	treatment	0	3.01E-02	0	0	3.01E-02	1./5E-04
Waste Plastic (styrene foam)	recycling/reuse	0	3.77E-03	0	0	3.77E-03	2.19E-05
Waste plastics from CR1 monitor	recycling/reuse	0	8.09E-02	0	0	8.09E-02	4.70E-04
Waste refractory	landfill	0	2.44E-03	0	0	2.44E-03	1.42E-05
Waste water treatment (WWT) sludge	landfill	0	3.43E-01	0	0	3.43E-01	0.20%
waste water treatment (WWT) sludge	recycling/reuse	0	3.72E-01	0	0	3.72E-01	0.22%
Wastepaper	recycling/reuse	0	8.34E-03	0	0	8.34E-03	4.84E-05
Wood, average	landfill	0	4.94E-03	0	0	4.94E-03	2.87E-05
Zinc (elemental)	landfill	9.01E-07	0	0	0	9.01E-07	5.23E-09
Total solid wast	e	9.55E+00	8.12E+01	8.33E+01	-1.66E+00	1.72E+02	100.00%

Table J-7. CRT solid waste outputs (kg/functional unit)

					0	/	
Material	Disposition	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Highly radioactive waste (Class C)	landfill	8.65e-06	0	0	0	8.65e-06	2.98e-03
Low-level radioactive waste	landfill	4.11e-04	1.38e-04	1.76e-03	1.76e-07	2.31e-03	79.47%
Radioactive waste (unspecified)	landfill	1.88e-05	0	0	0	1.88e-05	6.46e-03
Uranium, depleted	landfill	0	4.15e-05	5.27e-04	5.27e-08	5.69e-04	19.59%
Total radioactive was	te	4.39e-04	1.80e-04	2.28e-03	2.29e-07	2.90e-03	100.00%

Table J-8. CRT radioactive waste outputs (kg/functional unit)

Material	Disposition	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Americium-241 (isotope)	landfill	3.00e+02	0	0	0	3.00e+02	3.34e-06
Americium-243 (isotope)	landfill	6.53e+00	0	0	0	6.53e+00	7.26e-08
Antimony-124 (isotope)	surface water	6.75e-02	0	0	0	6.75e-02	7.52e-10
Antimony-124 (isotope)	treatment	0	6.21e-01	7.89e+00	7.89e-04	8.51e+00	9.47e-08
Antimony-125 (isotope)	treatment	0	2.47e+00	3.14e+01	3.14e-03	3.39e+01	3.77e-07
Argon-41 (isotope)	air	0	1.26e+03	1.60e+04	1.60e+00	1.72e+04	1.92e-04
Barium-140 (isotope)	treatment	0	4.60e-02	5.85e-01	5.85e-05	6.31e-01	7.02e-09
Bromine-89 (isotope)	air	0	1.45e-04	1.85e-03	1.85e-07	1.99e-03	2.22e-11
Bromine-90 (isotope)	air	0	5.91e-05	7.51e-04	7.52e-08	8.10e-04	9.02e-12
Carbon-14 (isotope)	air	2.97e+01	0	0	0	2.97e+01	3.31e-07
Cesium-134 (isotope)	air	1.14e-03	3.99e-03	5.07e-02	5.07e-06	5.58e-02	6.22e-10
Cesium-134 (isotope)	surface water	5.94e-02	0	0	0	5.94e-02	6.61e-10
Cesium-134 (isotope)	treatment	0	1.66e+00	2.11e+01	2.11e-03	2.28e+01	2.53e-07
Cesium-135 (isotope)	landfill	1.46e+05	0	0	0	1.46e+05	1.63e-03
Cesium-136 (isotope)	treatment	0	1.44e-02	9.03e-01	9.04e-05	9.18e-01	1.02e-08
Cesium-137 (isotope)	air	1.14e-03	3.01e-02	3.83e-01	3.83e-05	4.14e-01	4.61e-09
Cesium-137 (isotope)	landfill	4.09e-01	0	0	0	4.09e-01	4.55e-09
Cesium-137 (isotope)	surface water	8.71e-02	0	0	0	8.71e-02	9.69e-10
Cesium-137 (isotope)	treatment	0	2.49e+00	3.17e+01	3.17e-03	3.42e+01	3.80e-07
Chromium-51 (isotope)	air	0	7.89e-02	1.00e+00	1.00e-04	1.08e+00	1.20e-08
Chromium-51 (isotope)	treatment	0	2.99e+00	3.80e+01	3.80e-03	4.10e+01	4.56e-07
Cobalt-57 (isotope)	air	0	2.12e-04	2.69e-03	2.69e-07	2.90e-03	3.23e-11
Cobalt-57 (isotope)	treatment	0	7.25e-02	9.20e-01	9.21e-05	9.93e-01	1.11e-08
Cobalt-58 (isotope)	air	1.14e-03	5.49e+01	3.44e+03	3.44e-01	3.49e+03	3.89e-05
Cobalt-58 (isotope)	surface water	1.95e-01	0	0	0	1.95e-01	2.17e-09
Cobalt-58 (isotope)	treatment	0	2.95e+01	3.74e+02	3.75e-02	4.04e+02	4.50e-06
Cobalt-60 (isotope)	air	1.14e-03	2.04e-02	2.59e-01	2.59e-05	2.80e-01	3.12e-09
Cobalt-60 (isotope)	surface water	1.22e-01	0	0	0	1.22e-01	1.36e-09
Cobalt-80 (isotope)	treatment	0	7.74e+00	9.83e+01	9.83e-03	1.06e+02	1.18e-06
Curium-244 (isotope)	landfill	6.08e+02	0	0	0	6.08e+02	6.77e-06
Curium-245 (isotope)	landfill	6.78e-02	0	0	0	6.78e-02	7.55e-10
Iodine-129 (isotope)	landfill	9.58e-03	0	0	0	9.58e-03	1.07e-10
Iodine-131 (isotope)	air	6.67e-03	9.51e-02	1.21e+00	1.21e-04	1.31e+00	1.46e-08
Iodine-131 (isotope)	surface water	7.40e-03	0	0	0	7.40e-03	8.24e-11
Iodine-131 (isotope)	treatment	0	1.38e+00	1.75e+01	1.75e-03	1.89e+01	2.10e-07
Iodine-132 (isotope)	air	0	1.93e-02	2.45e-01	2.46e-05	2.65e-01	2.95e-09
Iodine-132 (isotope)	treatment	0	5.22e-01	6.64e+00	6.64e-04	7.16e+00	7.97e-08
Iodine-133 (isotope)	air	1.30e-02	8.82e+01	1.12e+03	1.12e-01	1.21e+03	1.35e-05
Iodine-133 (isotope)	treatment	0	5.91e-01	7.51e+00	7.52e-04	8.10e+00	9.02e-08
Iodine-134 (isotope)	air	0	1.00e-01	1.27e+00	1.27e-04	1.37e+00	1.53e-08
Iodine-135 (isotope)	air	0	5.03e-03	6.39e-02	6.39e-06	6.89e-02	7.67e-10
Iodine-135 (isotope)	treatment	0	4.24e-01	5.39e+00	5.39e-04	5.82e+00	6.47e-08
Iron-55 (isotope)	treatment	0	7.05e+00	8.95e+01	8.96e-03	9.66e+01	1.07e-06
Iron-59 (isotope)	treatment	0	3.62e-01	4.60e+00	4.60e-04	4.96e+00	5.52e-08
Krypton-85 (isotope)	aır	1.73e+02	2.08e+03	2.65e+04	2.65e+00	2.87e+04	3.20e-04
Krypton-85M (isotope)	aır	0	1.01e+02	1.28e+03	1.28e-01	1.38e+03	1.54e-05
Krypton-85M (isotope)	treatment	0	1.86e+00	2.37e+01	2.37e-03	2.55e+01	2.84e-07

Table J-9. CRT radioactivity outputs (Bq/functional unit)

Material	Disposition	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Krypton-87 (isotope)	air	0	3.76e+01	4.78e+02	4.78e-02	5.15e+02	5.74e-06
Krypton-88 (isotope)	air	0	1.76e+02	2.24e+03	2.24e-01	2.42e+03	2.69e-05
Lanthanum-140 (isotope)	treatment	0	4.93e-02	6.26e-01	6.26e-05	6.75e-01	7.52e-09
Lead-210 (isotope)	air	1.02e+00	0	0	0	1.02e+00	1.14e-08
Manganese-54 (isotope)	air	0	1.12e-03	1.42e-02	1.42e-06	1.53e-02	1.71e-10
Manganese-54 (isotope)	surface water	9.76e-03	0	0	0	9.76e-03	1.09e-10
Manganese-54 (isotope)	treatment	0	1.97e+00	2.50e+01	2.51e-03	2.70e+01	3.01e-07
Molybdenum-99 (isotope)	treatment	0	3.72e+06	4.73e+07	4.73e+03	5.10e+07	56.75%
Neptunium-237 (isotope)	landfill	9.39e+01	0	0	0	9.39e+01	1.05e-06
Niobium-95 (isotope)	air	0	4.45e-05	5.65e-04	5.65e-08	6.09e-04	6.78e-12
Niobium-95 (isotope)	treatment	0	5.08e-01	6.45e+00	6.45e-04	6.96e+00	7.74e-08
Palladium-107 (isotope)	landfill	3.30e-02	0	0	0	3.30e-02	3.67e-10
Plutonium-239 (isotope)	landfill	1.14e+05	0	0	0	1.14e+05	1.26e-03
Plutonium-240 (isotope)	landfill	1.62e+05	0	0	0	1.62e+05	1.80e-03
Plutonium-241 (isotope)	landfill	3.74e+07	0	0	0	3.74e+07	41.67%
Plutonium-242 (isotope)	landfill	6.10e+02	0	0	0	6.10e+02	6.79e-06
Polonium-210 (isotope)	air	1.76e+00	0	0	0	1.76e+00	1.96e-08
Potassium-40 (isotope)	air	2.73e-01	0	0	0	2.73e-01	3.04e-09
Protactinium-234 (isotope)	air	1.61e-02	0	0	0	1.61e-02	1.79e-10
Protactinium-234 (isotope)	surface water	2.98e-01	0	0	0	2.98e-01	3.32e-09
Radioactive aerosols and halogenes (unspecified)	air	8.95e-02	0	0	0	8.95e-02	9.96e-10
radioactive gas (unspecified)	air	2.86e+03	0	0	0	2.86e+03	3.18e-05
Radioactive substance (unspecified)	air	8.69e+02	9.19e+02	0	-2.38e+00	1.79e+03	1.99e-05
Radioactive substance (unspecified)	surface water	1.61e+01	8.52e+00	0	-2.21e-02	2.46e+01	2.73e-07
Radium-222 (isotope)	air	1.04e+01	0	0	0	1.04e+01	1.16e-07
Radium-224 (isotope)	surface water	9.67e-01	0	0	0	9.67e-01	1.08e-08
Radium-226 (isotope)	air	1.37e+00	0	0	0	1.37e+00	1.53e-08
Radium-226 (isotope)	landfill	6.35e+02	0	0	0	6.35e+02	7.07e-06
Radium-226 (isotope)	surface water	5.73e+02	0	0	0	5.73e+02	6.38e-06
Radium-228 (isotope)	air	1.35e-01	0	0	0	1.35e-01	1.50e-09
Radium-228 (isotope)	surface water	1.93e+00	0	0	0	1.93e+00	2.15e-08
Radon-220 (isotope)	air	3.17e+00	0	0	0	3.17e+00	3.53e-08
Radon-222 (isotope)	air	1.37e+05	0	0	0	1.37e+05	1.52e-03
Rubidium-88 (isotope)	air	0	4.13e-01	5.25e+00	5.25e-04	5.66e+00	6.30e-08
Ruthenium-103 (isotope)	treatment	0	6.21e-02	7.89e-01	7.89e-05	8.51e-01	9.47e-09
Samarium-151 (isotope)	landfill	1.35e+02	0	0	0	1.35e+02	1.50e-06
Selenium-79 (isotope)	landfill	1.05e-01	0	0	0	1.05e-01	1.17e-09
Silver-110M (isotope)	air	0	1.33e-06	1.68e-05	1.68e-09	1.82e-05	2.02e-13
Silver-110M (isotope)	surface water	2.93e-01	0	0	0	2.93e-01	3.26e-09
Silver-110M (isotope)	treatment	0	7.25e-01	9.20e+00	9.21e-04	9.93e+00	1.11e-07
Sodium-24 (isotope)	treatment	0	1.10e-01	1.40e+00	1.40e-04	1.51e+00	1.68e-08
Strontium-89 (isotope)	treatment	0	1.19e-01	1.52e+00	1.52e-04	1.63e+00	1.82e-08
Strontium-90 (isotope)	landfill	2.18e+04	0	0	0	2.18e+04	2.43e-04
Strontium-90 (isotope)	treatment	0	2.80e-02	3.56e-01	3.56e-05	3.84e-01	4.28e-09
Strontium-95 (isotope)	treatment	0	3.09e-01	3.93e+00	3.93e-04	4.23e+00	4.71e-08
Sulfur-136 (isotope)	treatment	0	6.65e-02	8.45e-01	8.46e-05	9.12e-01	1.01e-08

 Table J-9. CRT radioactivity outputs (Bq/functional unit)

Material	Disposition	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Technetium-99M (isotope)	air	0	5.96e-06	7.57e-05	7.58e-09	8.17e-05	9.10e-13
Technetium-99M (isotope)	landfill	4.46e+00	0	0	0	4.46e+00	4.97e-08
Technetium-99M (isotope)	treatment	0	4.33e-02	5.50e-01	5.50e-05	5.93e-01	6.60e-09
Thorium-228 (isotope)	air	1.14e-01	0	0	0	1.14e-01	1.27e-09
Thorium-228 (isotope)	surface water	3.87e+00	0	0	0	3.87e+00	4.30e-08
Thorium-230 (isotope)	air	2.33e-01	0	0	0	2.33e-01	2.59e-09
Thorium-230 (isotope)	landfill	6.35e+02	0	0	0	6.35e+02	7.07e-06
Thorium-230 (isotope)	surface water	2.79e+01	0	0	0	2.79e+01	3.10e-07
Thorium-232 (isotope)	air	7.28e-02	0	0	0	7.28e-02	8.10e-10
Thorium-234 (isotope)	air	1.61e-02	0	0	0	1.61e-02	1.79e-10
Thorium-234 (isotope)	surface water	2.98e-01	0	0	0	2.98e-01	3.32e-09
Tin-113 (isotope)	treatment	0	6.85e-02	8.70e-01	8.71e-05	9.39e-01	1.04e-08
Tin-126 (isotope)	landfill	1.84e-01	0	0	0	1.84e-01	2.05e-09
Tritium-3 (isotope)	air	3.47e+02	2.95e+03	3.74e+04	3.75e+00	4.07e+04	4.53e-04
Tritium-3 (isotope)	surface water	3.56e+03	0	0	0	3.56e+03	3.96e-05
Tritium-3 (isotope)	treatment	0	2.20e+04	2.80e+05	2.80e+01	3.02e+05	3.36e-03
Uranium-234 (isotope)	air	4.07e-01	0	0	0	4.07e-01	4.53e-09
Uranium-234 (isotope)	landfill	4.82e+02	0	0	0	4.82e+02	5.36e-06
Uranium-234 (isotope)	surface water	9.84e+00	0	0	0	9.84e+00	1.10e-07
Uranium-235 (isotope)	air	3.04e-03	0			3.04e-03	3.38e-11
Uranium-235 (isotope)	landfill	8.69e+00	0	0	0	8.69e+00	9.67e-08
Uranium-235 (isotope)	surface water	4.27e-01	0	0	0	4.27e-01	4.75e-09
Uranium-238 (isotope)	air	6.69e-01	0	0	0	6.69e-01	7.44e-09
Uranium-238 (isotope)	landfill	1.35e+02	0	0	0	1.35e+02	1.50e-06
Uranium-238 (isotope)	surface water	9.23e+00	0	0	0	9.23e+00	1.03e-07
Xenon-131M (isotope)	air	0	1.70e+02	2.16e+03	2.16e-01	2.33e+03	2.59e-05
Xenon-131M (isotope)	treatment	0	2.27e+01	2.88e+02	2.88e-02	3.11e+02	3.46e-06
Xenon-133 (isotope)	air	2.43e+03	6.28e+03	3.12e+05	3.12e+01	3.21e+05	3.57e-03
Xenon-133 (isotope)	treatment	0	3.48e+03	4.43e+04	4.43e+00	4.78e+04	5.31e-04
Xenon-133M (isotope)	air	0	1.99e+04	2.07e+04	2.07e+00	4.06e+04	4.52e-04
Xenon-133M (isotope)	treatment	0	2.85e+01	3.62e+02	3.63e-02	3.91e+02	4.35e-06
Xenon-135 (isotope)	air	0	9.27e+02	1.18e+04	1.18e+00	1.27e+04	1.41e-04
Xenon-135 (isotope)	treatment	0	2.60e+01	3.30e+02	3.30e-02	3.56e+02	3.96e-06
Xenon-135M (isotope)	air	0	1.77e+01	2.25e+02	2.25e-02	2.42e+02	2.70e-06
Xenon-138 (isotope)	air	0	5.87e+01	7.45e+02	7.45e-02	8.04e+02	8.95e-06
Zinc-85 (isotope)	treatment	0	3.33e-02	4.23e-01	4.23e-05	4.56e-01	5.07e-09
Zirconium-93 (isotope)	landfill	5.87e-01	0	0	0	5.87e-01	6.53e-09
Zirconium-95 (isotope)	air	0	1.15e-04	1.46e-03	1.46e-07	1.57e-03	1.75e-11
Total radioactivity outpu	uts	3.80e+07	3.78e+06	4.80e+07	4.80e+03	8.98e+07	100.00%

Table J-9. CRT radioactivity outputs (Bq/functional unit)

Material	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
1,4-butanolide	0	4.06e-04	0	0	4.06E-04	1.12E-06
15" LCD light guide	0	3.74e-01	0	0	3.74E-01	1.03E-03
1-methyl-2-pyrrolidinone	0	4.06e-04	0	0	4.06E-04	1.12E-06
2-(2-butoxyethoxy)-ethanol acetate	0	8.08e-06	0	0	8.08E-06	2.23E-08
3,4,5-trifluorobromobenzene	0	2.64e-04	0	0	2.64E-04	7.29E-07
3,4-difluorobromobenzene	0	3.65e-04	0	0	3.65E-04	1.01E-06
4-4(-propylcyclohexyl)cyclohexanone	0	2.18e-04	0	0	2.18E-04	6.00E-07
4-bromophenol	0	3.27e-04	0	0	3.27E-04	9.00E-07
4-ethylphenol	0	7.00e-05	0	0	7.00E-05	1.93E-07
4-pentylphenol	0	3.42e-04	0	0	3.42E-04	9.43E-07
4-propionylphenol	0	1.94e-04	0	0	1.94E-04	5.36E-07
AlNd	0	2.97e-05	0	0	2.97E-05	8.18E-08
Aluminum (elemental)	0	1.34e-01	0	0	1.34E-01	3.70E-04
Argon	0	3.53e-05	0	0	3.53E-05	9.74E-08
Assembled 15" LCD backlight unit	0	1.48e+00	0	0	1.48E+00	4.07E-03
Assembled LCD monitor	0	0	6.50E+00	0	6.50E+00	1.79%
Backlight lamp (CCFL)	0	1.94e-03	0	0	1.94E-03	5.34E-06
Barium Carbonate	0	1.37e-02	0	0	1.37E-02	3.79E-05
Bauxite (Al2O3, ore)	0	5.09e-01	0	0	5.09E-01	1.40E-03
Cables/wires	0	2.34e-01	0	0	2.34E-01	6.45E-04
Coal, average (in ground)	1.72E+00	8.03E+00	6.69E+01	1.27E-02	7.67E+01	21.15%
Fuel oil #4	0	0	0	-6.18E-02	-6.18E-02	-1.70E-04
Glass, unspecified	0	4.37E-02	0	0	4.37E-02	1.20E-04
Glycol ethers	0	4.06E-04	0	0	4.06E-04	1.12E-06
Indium tin oxide	0	5.26E-04	0	0	5.26E-04	1.45E-06
Iron (Fe. ore)	3.26E+00	0	0	0	3.26E+00	8.98E-03
Iron scrap	4 63E-01	0	0	0	4 63E-01	1.28E-03
LCD front glass (with color filters)	0	1 78E-01	0	0	1.78E-01	4 92E-04
LCD glass	0	4 52E-01	0	0	4 52E-01	1.25E-03
LCD material (confidential)	0	3 11E-04	0	0	3 11E-04	8 56E-07
LCD module	0	1 18E+00	0	0	1 18E+00	3 26E-03
LCD snacers unspecified	0	1 69E-05	0	0	1 69E-05	4 66E-08
Liquid crystals for 15" LCD	0	1.24E-03	0	0	1.03E 03	3 43E-06
Mercury	0	3.99E-06	0	0	3 99E-06	1 10E-08
Metals remaining unspeciated	0	6.81E-04	0	0	6.81E-04	1.88E-06
Mild fiber	0	7 34E-07	0	0	7 34F-07	2.02E-09
Molybdenum	0	1 78E-04	0	0	1 78F-04	4 92E-07
MoW	0	9.09E-04	0	0	9.09E-04	2.51E-06
Natural gas	0	4 22E+00	5 22E+00	-5 75E-02	9.30E+00	2.5112-00
Natural gas	2 29E+02	5.16E+00	5.22E+00	-1.08E+00	2 33E+02	64 25%
Neon	2.2)E+02	6 31E 05	0	-1.082+00	6.31E.05	1 74E 07
Potroloum (in ground)	7.00E.01	0.51E-05	1 42E±00	1.00E+00	2.24E±01	6 45%
Pictroleum (m ground)	7.09E-01	2.23E+01	1.42E+00	-1.00E+00	2.54E+01	1.02E.04
Palaninan	0	3.72E-02	0	0	3.72E-02	1.12E.04
F Utdi 1251 Doly(mothyd motheografiato)	0	4.07E-02	0	0	4.07E-02	1.12E-04
Polyeerhonata rasin	0	5.14E.01	0	0	5.14E.01	1.00E-03
Polyesten odhasiya	0	5.10E-01	0	0	5.10E-01	1.42E-03
Polyester adnesive	0	0.25E-04	0	0	0.25E-04	1./2E-06
roryemytene tereprinatate	0	5.88E-02	0	0	5.88E-02	1.62E-04

Table J-10. LCD primary material inputs (kg/functional unit)

Material		Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Polyimide alignment layer, unspec	cified	0	4.86E-04	0	0	4.86E-04	1.34E-06
Polyvinyl alcohol		0	8.61E-03	0	0	8.61E-03	2.37E-05
Potassium Carbonate		0	1.75E-02	0	0	1.75E-02	4.83E-05
PPE		0	3.00E-01	0	0	3.00E-01	8.27E-04
Printed wiring board (PWB)		0	3.74E-01	0	0	3.74E-01	1.03E-03
PWB-laminate		0	3.74E-01	0	0	3.74E-01	1.03E-03
Recycled LCD glass		0	9.54E-02	0	0	9.54E-02	2.63E-04
Rubber, unspecified		0	6.01E-04	0	0	6.01E-04	1.66E-06
Sand		0	1.11E-01	0	0	1.11E-01	3.07E-04
Sodium Carbonate		0	2.26E-02	0	0	2.26E-02	6.23E-05
Solder (60% tin, 40% lead)		0	3.81E-02	0	0	3.81E-02	1.05E-04
Solder (63% tin; 37% lead)		0	2.24E-02	0	0	2.24E-02	6.18E-05
Steel		0	2.53E+00	0	0	2.53E+00	6.97E-03
Strontium Carbonate		0	1.53E-02	0	0	1.53E-02	4.23E-05
Styrene-butadiene copolymers		0	3.62E-01	0	0	3.62E-01	9.97E-04
Titanium		0	1.33E-04	0	0	1.33E-04	3.67E-07
Triallyl isocyanurate		0	1.54E-05	0	0	1.54E-05	4.26E-08
Triphenyl phosphate		0	9.25E-02	0	0	9.25E-02	2.55E-04
Unspecified LCD material		0	1.19E-04	0	0	1.19E-04	3.29E-07
Uranium, yellowcake		0	1.03E-03	1.81E-03	3.4352E-07	2.84E-03	7.84E-06
Zircon Sand		0	1.31E-03	0	0	1.31E-03	3.62E-06
	Total primary inputs	2.35E+02	4.97E+01	8.01E+01	-2.19E+00	3.63E+02	100.00%

Table J-10. LCD primary material inputs (kg/functional unit)

Material	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
1,4-butanolide	0	4.04e-05	0	0	4.04e-05	1.94e-07
1-Methoxy-2-propanol	0	1.10e-02	0	0	1.10e-02	5.27e-05
2-(2-butoxyethoxy)-ethanol	0	3.04e-02	0	0	3.04e-02	1.46e-04
2,2,4-trimethylpentane	0	1.52e-05	0	0	1.52e-05	7.32e-08
2-ethoxyl ethylacetate	0	1.78e-03	0	0	1.78e-03	8.53e-06
Acetic acid	0	6.40e-03	0	0	6.40e-03	3.07e-05
Acetone	0	1.56e-01	0	0	1.56e-01	7.50e-04
Al-etchant, unspecified	0	5.88e-03	0	0	5.88e-03	2.83e-05
Aluminum Oxide	0	1.56e-03	0	0	1.56e-03	7.50e-06
Aluminum sulfate	3.63e-04	1.05e-01	0	0	1.05e-01	5.04e-04
Ammonia	0	1.55e-02	0	0	1.55e-02	7.46e-05
Ammonium bifluoride	0	2.36e-03	0	0	2.36e-03	1.13e-05
Ammonium chloride	0	3.42e-02	0	0	3.42e-02	1.64e-04
Ammonium fluoride	0	1.14e-02	0	0	1.14e-02	5.48e-05
Ammonium hydroxide	0	3.42e-02	0	0	3.42e-02	1.64e-04
Argon	0	7.87e-03	0	0	7.87e-03	3.78e-05
Barium sulfate	9.64e-04	0	0	0	9.64e-04	4.63e-06
Bauxite	2.10e-04	0	0	0	2.10e-04	1.01e-06
Bauxite (Al2O3, ore)	5.59e-04	2.16e-03	0	-7.63e-05	2.64e-03	1.27e-05
Bentonite (in ground)	1.00e-03	0	0	0	1.00e-03	4.82e-06
Borax	0	9.13e-05	0	0	9.13e-05	4.39e-07
Calcium hydroxide	0	1.39e-01	0	0	1.39e-01	6.67e-04
Calcium sulfate	7.37e-06	0	0	0	7.37e-06	3.54e-08
Calcium sulfate (CaSO4, ore)	9.75e-06	0	0	0	9.75e-06	4.68e-08
Carbon dioxide	0	4.86e-03	0	0	4.86e-03	2.33e-05
Cerium Oxide	0	1.52e-04	0	0	1.52e-04	7.29e-07
Chlorine	0	1.55e-02	0	0	1.55e-02	7 46e-05
Chromium ore	1.79e-08	0	0	0	1.79e-08	8.59e-11
Chromium Oxide	0	2.60e-06	0	0	2.60e-06	1.25e-08
Clay (in ground)	1.30e-03	0	0	1.69e+00	1.69e+00	8.12e-03
Cleaner, unspecified	0	1.47e-04	0	0	1.47e-04	7.05e-07
Coal. average (in ground)	0	0	0	7.38e-04	7.38e-04	3.54e-06
Copper (Cu. ore)	2.63e-04	0	0	0	2.63e-04	1.26e-06
Cresol-formaldehvde resin	0	8.29e-04	0	0	8.29e-04	3.98e-06
Cr-etchant, unspecified	0	1.77e-02	0	0	1.77e-02	8.49e-05
Cyclohexane	0	3.91e-03	0	0	3.91e-03	1.88e-05
Deoiling agent (unspecified)	6 53e-04	0	0	0	6 53e-04	3 14e-06
Developing solution unspecified	0.550 01	4 00e-02	0	0	4 00e-02	1 92e-04
Diethyl ether	0	9.28e-05	0	0	9.28e-05	4 46e-07
Diluent unspecified	0	8 27e-03	0	0	8 27e-03	3.97e-05
Dimethylsulfoxide	0	6.63e-02	0	0	6.63e-02	3 18e-04
Dioctyl Sebacate	1 14e-04	0.050 02	0	0	1 14e-04	5.47e-07
Dolomite (in ground)	2.08e-03	0	0	0	2 08e-03	9.98e-06
Ethanol	2.000-05	2 53e-02	0	0	2.08e-03	1 21e-04
Ethanol amine	0	7.85e-02	0	0	7.85e-02	3 77e-04
Ethylacetate	0	9.680.04	0	0	9.682.04	4 650 06
Etary Nanhtol Sulnhonic Acid (ENSA)	4 880-05	0.000-04	0	0	4 880-05	2 34e 07
Explicit Suprior Suprome Acta (ENSA)	000-03	1 430 02	0	0	1.430.02	6 860 05
Explosive (unspecified)	0 770 05	1.430-02	0	0	0.770.05	4.600.07
Explosive (unspecified)	9.77e-05	U	0	U	9.776-05	4.096-0/

Table J-11. LCD ancillary material inputs (kg/functional unit)

Material	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Ferric chloride	0	8.92e-03	0	0	8.92e-03	4.28e-05
Ferromanganese (Fe, Mn, C)	3.83e-07	0	0	0	3.83e-07	1.84e-09
Fluorocarbon resin	0	3.38e-06	0	0	3.38e-06	1.63e-08
Fluorspar (CaF2, ore)	5.37e-06	0	0	0	5.37e-06	2.58e-08
Flux, unspecified	0	7.35e-05	0	0	7.35e-05	3.53e-07
Formaldehyde	0	2.91e-03	0	0	2.91e-03	1.40e-05
Glycol ethers	0	3.16e-02	0	0	3.16e-02	1.52e-04
Gravel	9.08e-08	0	0	0	9.08e-08	4.36e-10
Gravel/Sand	1.01e-03	0	0	0	1.01e-03	4.85e-06
HCFC-225ca	0	1.37e-04	0	0	1.37e-04	6.57e-07
HCFC-225cb	0	1.37e-04	0	0	1.37e-04	6.57e-07
Helium	0	6.18e-04	0	0	6.18e-04	2.97e-06
Heptane	0	1.03e-02	0	0	1.03e-02	4.93e-05
Hexamethyldisilizane	0	2.58e-04	0	0	2.58e-04	1.24e-06
Hydrochloric acid	0	1.29e-01	0	0	1.29e-01	6.21e-04
Hydrofluoric acid	0	4.58e-02	0	0	4.58e-02	2.20e-04
Hydrogen	0	4.44e-01	0	0	4.44e-01	2.13e-03
Hydrogen peroxide	0	1.38e-02	0	0	1.38e-02	6.65e-05
Iron (Fe, ore)	7.65e-03	0	0	7.04e-04	8.35e-03	4.01e-05
Iron sulfate (FeSO4, ore)	7.99e-06	0	0	0	7.99e-06	3.84e-08
Isopropyl alcohol	0	3.49e-01	0	0	3.49e-01	1.67e-03
ITO etchant, unspecified	0	2.94e-03	0	0	2.94e-03	1.41e-05
Krypton	0	2.58e-05	0	0	2.58e-05	1.24e-07
Lead (Pb, ore)	2.47e-05	0	0	0	2.47e-05	1.19e-07
Lime	0	4.74e-02	3.95e-01	7.49e-05	4.42e-01	2.12e-03
Limestone	0	1.08e-01	8.99e-01	1.71e-04	1.01e+00	4.84e-03
Limestone (CaCO3, in ground)	5.07e-01	5.49e-02	0	-1.55e-01	4.06e-01	1.95e-03
LNG	0	1.94e+02	0	0	1.94e+02	9.32e-01
Lubricant (unspecified)	2.02e-02	0	0	0	2.02e-02	9.68e-05
Manganese (Mn, ore)	1.04e-08	0	0	0	1.04e-08	5.00e-11
Methyl ethyl ketone	0	4.91e-04	0	0	4.91e-04	2.36e-06
Monosilane	0	1.12e-03	0	0	1.12e-03	5.40e-06
Natural gas (in ground)	0	0	0	1.09e-03	1.09e-03	5.22e-06
N-Butylacetate	0	3.83e-02	0	0	3.83e-02	1.84e-04
Nickel (Ni, ore)	6.05e-09	0	0	0	6.05e-09	2.91e-11
Nitric acid	0	7.23e-02	0	0	7.23e-02	3.47e-04
Nitric acid second cerium ammonium	0	1.13e-02	0	0	1.13e-02	5.43e-05
Nitrogen	0	6.02e+00	0	0	6.02e+00	2.89e-02
Nitrogen fluoride	0	1.08e-01	0	0	1.08e-01	5.19e-04
Nitrous oxide	0	1.36e-03	0	0	1.36e-03	6.51e-06
Oil (in ground)	0	0	0	8.06e-03	8.06e-03	3.87e-05
Olivine	2.72e-07	0	0	0	2.72e-07	1.31e-09
Olivine ore	7.92e-06	0	0	0	7.92e-06	3.80e-08
Orthoboric acid	0	7.30e-04	0	0	7.30e-04	3.51e-06
Oxygen	0	7.75e-03	0	0	7.75e-03	3.72e-05
Perchloric acid	0	3.82e-03	0	0	3.82e-03	1.84e-05
Perfluoromethane	0	1.29e-03	0	0	1.29e-03	6.18e-06
Phenolsulphonic Acid	1.53e-03	0	0	0	1.53e-03	7.34e-06
Phosphine	0	2.69e-02	0	0	2.69e-02	1.29e-04

Table J-11. LCD ancillary material inputs (kg/functional unit)

Material	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Phosphoric acid	0	3.95e-02	0	0	3.95e-02	1.90e-04
Photoresist, unspecified	0	1.68e-02	0	0	1.68e-02	8.05e-05
Polyaluminum chloride	0	6.40e-03	0	0	6.40e-03	3.07e-05
Polyethylene glycol	0	2.23e-02	0	0	2.23e-02	1.07e-04
Polyethylene mono(nonylphenyl) ether glycol	0	3.40e-04	0	0	3.40e-04	1.63e-06
Polyethylene terephthalate	0	3.20e-02	0	0	3.20e-02	1.53e-04
Polyimide, unspecified	0	2.94e-05	0	0	2.94e-05	1.41e-07
Potassium chloride (KCl, as K2O, in ground)	3.03e-05	0	0	0	3.03e-05	1.45e-07
Potassium hydroxide	0	1.88e-02	0	0	1.88e-02	9.04e-05
Potassium permanganate	0	5.14e-04	0	0	5.14e-04	2.47e-06
Potassium peroxymonosulfate	0	3.12e-02	0	0	3.12e-02	1.50e-04
Process material for backlight assembly	0	7.03e-05	0	0	7.03e-05	3.37e-07
Propylene glycol	0	4.46e-03	0	0	4.46e-03	2.14e-05
Propylene glycol monomethyl ether acetate	0	1.56e-02	0	0	1.56e-02	7.50e-05
Pumice	0	3.64e-03	0	0	3.64e-03	1.75e-05
PWB-solder mask solids	0	1.93e-02	0	0	1.93e-02	9.25e-05
Pyrite (FeS2, ore)	3.77e-02	0	0	0	3.77e-02	1.81e-04
Raw materials (unspecified)	4.06e-03	0	0	0	4.06e-03	1.95e-05
Rinse, unspecified	0	5.27e-02	0	0	5.27e-02	2.53e-04
Sand (in ground)	9.55e-03	1.32e-03	0	5.60e-01	5.71e-01	2.74e-03
Silver (Ag, ore)	4.51e-10	0	0	0	4.51e-10	2.17e-12
Sodium Carbonate	0	1.42e-02	0	0	1.42e-02	6.82e-05
Sodium chloride (NaCl, in ground or in sea)	4.37e-01	6.08e-04	0	1.08e-05	4.38e-01	2.10e-03
Sodium Dichromate	1.29e-04	0	0	0	1.29e-04	6.19e-07
Sodium dihydrogen phosphate dihydrate	0	4.06e-06	0	0	4.06e-06	1.95e-08
Sodium hydroxide	0	4.45e-01	0	0	4.45e-01	2.14e-03
Solder, unspecified	0	7.35e-05	0	0	7.35e-05	3.53e-07
Sulfur (S, in ground)	1.53e-02	0	0	0	1.53e-02	7.34e-05
Sulfur hexafluoride	0	1.62e-02	0	0	1.62e-02	7.79e-05
Sulfuric acid	0	3.25e-01	0	0	3.25e-01	1.56e-03
Surfactant, unspecified	0	1.09e-04	0	0	1.09e-04	5.25e-07
Synthetic resin, unspecified	0	6.57e-04	0	0	6.57e-04	3.16e-06
Tetrahydrofuran	0	3.82e-03	0	0	3.82e-03	1.84e-05
Tetramethyl ammonium hydroxide	0	1.29e-01	0	0	1.29e-01	6.18e-04
Tin (Sn, ore)	1.19e-02	0	0	0	1.19e-02	5.72e-05
Toluene	0	2.75e-02	0	0	2.75e-02	1.32e-04
Unspecified ancillary material	0	4.19e-04	0	0	4.19e-04	2.01e-06
Unspecified LCD process material	0	2.57e-02	0	0	2.57e-02	1.23e-04
Unspecified LCD process material ("Natural						
Sweeper")	0	1.09e-04	0	0	1.09e-04	5.23e-07
Uranium (U, ore)	0	0	0	1.00e-08	1.00e-08	4.81e-11
Water	0	6.88e-02	0	0	6.88e-02	3.30e-04
Xylene (mixed isomers)	0	1.57e-03	0	0	1.57e-03	7.54e-06
Zinc (Zn, ore)	6.61e-10	0	0	0	6.61e-10	3.18e-12
Total ancillary inputs	1.06e+00	2.04e+02	1.29e+00	2.11e+00	2.08e+02	100.00%
APPENDIX J

	Table J-12. LCD utility inputs										
Material	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total					
Fuels (kg/functional unit):											
Fuel Oil #2 (distillate)	0	5.42E-02	0	0	5.42E-02	0.14%					
Fuel oil #4 (avg. of #2 + #6)	0	2.11E-01	0	-9.09E-01	-6.99E-01	-1.81%					
Fuel oil #6 (residual)	0	1.25E-01	0	0	1.25E-01	0.33%					
Kerosene	0	4.65E-01	0	0	4.65E-01	1.21%					
LNG	0	3.22E+00	0	0	3.22E+00	8.34%					
LPG	0	1.68E+01	0	1.38E-03	1.68E+01	43.63%					
Natural gas	0	1.16E+00	0	-8.61E-01	3.01E-01	0.78%					
Steam	0	1.45E-01	0	0	1.45E-01	0.37%					
Coal, average (in ground)	2.49E+00	6.86E-01	0	-7.66E-03	3.17E+00	8.21%					
Coal, lignite (in ground)	4.10E-01	0	0	0	4.10E-01	1.06%					
Natural gas (in ground)	1.03E+01	2.41E+00	0	-1.38E-01	1.26E+01	32.66%					
Petroleum (in ground)	1.52E+00	4.84E-01	0	-3.81E-02	1.96E+00	5.08%					
Uranium (U, ore)	7.86E-05	1.15E-05	0	-1.32E-07	9.01E-05	2.33E-06					
Total fuels	s 1.47E+01	2.58E+01	0	-1.95E+00	3.86E+01	100.00%					
Electricity (MJ/functional unit):											
Electricity	3.46E+01	3.16E+02	8.53E+02	1.62E-01	1.20E+03						
Water (kg or L/functional unit):											
Water	2.63E+02	2.15E+03	4.25E+02	-1.80E+01	2.82E+03						
Total energy (fuels and electricity, MJ/functi	ional unit):										
Energy	6.33E+02	1.44E+03	8.53E+02	-8.44E+01	2.84E+03						

Material	Disposition	Upstream	Mfg	Use	EOL	Total	Fraction of Total
1,1,1-Trichloroethane	Air	1.62e-08	2.28e-07	7.16e-07	-1.95e-08	9.40e-07	2.72e-09
1,2-Dichloroethane	Air	3.24e-08	1.75e-07	1.34e-06	-3.85e-08	1.51e-06	4.36e-09
1,2-Dichlorotetrafluoroethane	Air	8.90e-08	0	0	0	8.90e-08	2.57e-10
1,4-Dichlorobenzene	Air	1.89e-07	1.89e-08	0	-1.49e-09	2.06e-07	5.96e-10
2,3,7,8-TCDD	Air	0	5.90e-14	4.79e-13	9.08e-17	5.38e-13	1.55e-15
2,3,7,8-TCDF	Air	0	2.05e-13	1.71e-12	3.24e-16	1.91e-12	5.53e-15
2,4-Dinitrotoluene	Air	2.27e-10	1.22e-09	9.37e-09	-2.74e-10	1.05e-08	3.05e-11
2-Chloroacetophenone	Air	5.67e-09	3.06e-08	2.34e-07	-9.35e-11	2.70e-07	7.82e-10
2-Methylnaphthalene	Air	3.78e-09	1.18e-09	9.95e-10	-3.53e-11	5.92e-09	1.71e-11
3-Methylcholanthrene	Air	2.83e-10	2.84e-11	0	-2.66e-12	3.09e-10	8.94e-13
5-Methyl chrysene	Air	1.78e-11	9.60e-11	7.36e-10	-2.15e-11	8.29e-10	2.40e-12
Acenaphthene	Air	9.25e-10	1.49e-08	2.12e-08	-5.04e-10	3.65e-08	1.06e-10
Acenaphthylene	Air	4.89e-10	1.27e-09	8.42e-09	-2.47e-10	9.93e-09	2.87e-11
Acetaldehyde	Air	7.90e-07	2.49e-06	1.91e-05	-5.57e-07	2.18e-05	6.30e-08
Acetic acid	Air	4.22e-06	1.36e-03	0	0	1.36e-03	3.93e-06
Acetone	Air	3.26e-07	1.86e-04	0	0	1.86e-04	5.39e-07
Acetophenone	Air	1.22e-08	6.55e-08	5.02e-07	-1.47e-08	5.65e-07	1.63e-09
Acetylene	Air	1.14e-05	0	0	0	1.14e-05	3.30e-08
Acrolein	Air	2.35e-07	1.27e-06	9.71e-06	-2.83e-07	1.09e-05	3.16e-08
Adsorbable organic halides	Air	3.56e-17	0	0	0	3.56e-17	1.03e-19
Alcohols	Air	2.85e-06	0	0	0	2.85e-06	8.23e-09
Aldehydes	Air	8.53e-04	8.98e-05	0	-1.05e-04	8.37e-04	2.42e-06
Al-etchant, unspecified	Air	0	1.37e-02	0	0	1.37e-02	3.97e-05
Alkane (unspecified)	Air	1.07e-04	0	0	0	1.07e-04	3.10e-07
Alkenes	Air	1.79e-05	0	0	0	1.79e-05	5.18e-08
Alkyne (unspecified)	Air	5.46e-08	0	0	0	5.46e-08	1.58e-10
Aluminum (elemental)	Air	8.09e-05	9.59e-07	0	-5.72e-08	8.18e-05	2.36e-07
Ammonia	Air	1.12e-02	6.26e-02	0	-6.95e-05	7.37e-02	2.13e-04
Anthracene	Air	5.61e-10	1.68e-09	7.27e-09	-2.09e-10	9.30e-09	2.69e-11
Antimony	Air	2.28e-07	3.24e-06	1.63e-06	-4.25e-08	5.05e-06	1.46e-08
Argon	Air	0	5.80e-03	0	0	5.80e-03	1.68e-05
Aromatic hydrocarbons	Air	7.67e-05	2.55e-09	0	-9.05e-11	7.67e-05	2.22e-07
Arsenic	Air	1.63e-06	3.14e-06	1.40e-05	-1.72e-06	1.70e-05	4.93e-08
Barium	Air	1.39e-06	1.76e-06	7.67e-07	-1.65e-08	3.91e-06	1.13e-08
Benzaldehyde	Air	2.78e-14	0	0	0	2.78e-14	8.04e-17
Benzene	Air	8.85e-02	2.70e-03	4.36e-05	-4.82e-04	9.07e-02	2.62e-04
Benzo[a]anthracene	Air	3.92e-10	2.78e-09	3.46e-09	-8.13e-11	6.55e-09	1.89e-11
Benzo[a]pyrene	Air	4.40e-08	1.91e-10	1.27e-09	-3.93e-11	4.54e-08	1.31e-10
Benzo[b,j,k]fluoranthene	Air	8.92e-11	1.37e-09	3.97e-09	-1.07e-10	5.32e-09	1.54e-11
Benzo[b]fluoranthene	Air	2.91e-10	3.09e-11	0	-2.79e-12	3.19e-10	9.24e-13
Benzo[g,h,i]perylene	Air	2.23e-10	1.48e-09	1.35e-09	-2.83e-11	3.02e-09	8.74e-12
Benzo[k]fluoranthene	Air	2.91e-10	3.09e-11	0	-2.79e-12	3.19e-10	9.24e-13
Benzyl chloride	Air	5.67e-07	3.05e-06	2.34e-05	-6.84e-07	2.64e-05	7.62e-08
Beryllium	Air	2.19e-07	1.86e-07	7.12e-07	-1.99e-07	9.18e-07	2.65e-09
Biphenyl	Air	1.38e-09	7.42e-09	5.69e-08	-1.66e-09	6.40e-08	1.85e-10
Boron	Air	1.79e-05	0	0	0	1.79e-05	5.16e-08
Bromine	Air	9.94e-07	0	0	0	9.94e-07	2.87e-09
Bromoform	Air	3.16e-08	1.70e-07	1.31e-06	-3.81e-08	1.47e-06	4.25e-09
Bromomethane	Air	1.30e-07	6.98e-07	5.36e-06	-1.56e-07	6.03e-06	1.74e-08

 Table J-13. LCD air emissions (kg/functional unit)

Material	Disposition	Upstream	Mfg	Use	EOL	Total	Fraction of Total
Butane	Air	3 42e-04	3 31e-05	0	-3 11e-06	3 72e-04	1 07e-06
Butene	Air	1.09e-07	0	0	0	1.09e-07	3 15e-10
Cadmium	Air	2.93e-07	5 25e-07	1.80e-06	-6 58e-08	2.55e-06	7 37e-09
Calcium	Air	8 74e-05	8 31e-07	0	-4 96e-08	8 82e-05	2 55e-07
Carbon dioxide	Air	1.07e+02	6.22e+01	1 66e+02	1 39e+00	3 36e+02	9 72e-01
Carbon disulfide	Air	1.07 c +02	5.67e-07	4 35e-06	-1 27e-07	4 90e-06	1 42e-08
Carbon monoxide	Air	3 74e-01	3 85e-02	3.02e-02	-3 45e-03	4 39e-01	1 27e-03
Carbon tetrachloride	Air	0	0	0	5 85e-10	5 85e-10	1.69e-12
Chloride ions	Air	1 77e-04	2.09e-04	6 78e-05	-1 64e-06	4 52e-04	1 31e-06
Chlorine	Air	6 64e-07	1.01e-09	0	-1 80e-10	6 65e-07	1.92e-09
Chloroacetophenone	Air	0	0	0	-6 74e-09	-6 74e-09	-1 95e-11
Chlorobenzene	Air	1 78e-08	9 60e-08	7 36e-07	-2.15e-08	8 29e-07	2.40e-09
Chloroform	Air	4 78e-08	2.57e-07	1.97e-06	-5 70e-08	2.22e-06	6 43e-09
Chromium	Air	6 40e-09	0	0	0	6 40e-09	1.85e-11
Chromium (III)	Air	1 50e-06	2.96e-06	9.07e-06	-1 43e-06	1.21e-05	3 50e-08
Chromium (VI)	Air	1 50e-06	1 52e-06	2.69e-06	-1 43e-06	4 29e-06	1 24e-08
Chrysene	Air	3 77e-10	1.89e-09	3.81e-09	-1 00e-10	5 98e-09	1.2.10 00
Cobalt	Air	5 36e-07	4 13e-06	4 54e-06	-1 30e-07	9.07e-06	2.62e-08
Copper	Air	8 26e-07	1.15e-06	3 72e-07	-1 65e-08	2.33e-06	6 75e-09
Cr-etchant unspecified	Air	0	4 12e-02	0	0	4 12e-02	1 19e-04
Cumene	Air	4 30e-09	2 75e-09	1 77e-07	-5 18e-09	1.72e-02	5 18e-10
Cumene hydroperoxide	Air	0	2.04e-08	0	0	2.04e-08	5 89e-11
Cyanide (-1)	Air	2.04e-06	1.09e-05	8 37e-05	-2.44e-06	9 42e-05	2.72e-07
Cyclohexane	Air	0	4 85e-05	0	0	4 85e-05	1 40e-07
Di(2-ethylhexyl)phthalate	Air	5 92e-08	3 19e-07	2.44e-06	-7 13e-08	2.75e-06	7 95e-09
Dibenzo[a_h]anthracene	Air	1.98e-10	1.02e-09	3.26e-10	-1.85e-12	1.54e-09	4.46e-12
Dichlorobenzene (mixed isomers)	Air	0	0	0	-2.82e-10	-2.82e-10	-8 16e-13
Dichloromethane	Air	2.35e-07	1 27e-06	9 71e-06	-2.82e-07	1.09e-05	3 16e-08
Diethyl ether	Air	0	9.26e-05	0	0	9 26e-05	2.68e-07
Diethylene glycol	Air	0	9.69e-05	0	0	9.69e-05	2.80e-07
Dimethyl sulfate	Air	3.89e-08	2.09e-07	1.61e-06	-4.69e-08	1.81e-06	5.23e-09
Dimethylbenzanthracene	Air	2.36e-09	2.36e-10	0	-2.23e-11	2.58e-09	7.45e-12
Dioxins remaining unspeciated	Air	1 32e-11	8 22e-12	2.18e-11	1 33e-10	1 76e-10	5 08e-13
Ethane	Air	6.08e-04	4.88e-05	0	-4.59e-06	6.53e-04	1.89e-06
Ethanethiol	Air	4.50e-07	0	0	0	4.50e-07	1.30e-09
Ethanol	Air	6.51e-07	4.63e-05	0	0	4.69e-05	1.36e-07
Ethvl Chloride	Air	3.40e-08	1.83e-07	1.41e-06	-4.10e-08	1.58e-06	4.57e-09
Ethylacetate	Air	0	1.22e-07	0	0	1.22e-07	3.53e-10
Ethylacetate	Treatment	0	2.32e-06	0	0	2.32e-06	6.70e-09
Ethylbenzene	Air	2.28e-07	4.48e-07	3.16e-06	-9.00e-08	3.75e-06	1.08e-08
Ethylene	Air	1.02e-04	0	0	0	1.02e-04	2.94e-07
Ethylene dibromide	Air	9.73e-10	5.24e-09	4.02e-08	-1.17e-09	4.52e-08	1.31e-10
Fluoranthene	Air	1.07e-09	6.31e-09	2.50e-08	-6.98e-10	3.17e-08	9.17e-11
Fluorene	Air	1.20e-09	6.68e-09	3.13e-08	-8.93e-10	3.83e-08	1.11e-10
Fluoride	Air	0	3.78e-08	7.29e-06	1.38e-09	7.33e-06	2.12e-08
Fluorides (F-)	Air	9.48e-07	2.24e-05	0	-2.05e-07	2.32e-05	6.69e-08
Fluorine	Air	2.83e-07	0	0	0	2.83e-07	8.18e-10
Formaldehyde	Air	1.87e-05	1.09e-04	3.16e-05	-5.78e-06	1.54e-04	4.45e-07
Furans, remaining unspeciated	Air	6.10e-11	3.02e-11	3.48e-11	-7.40e-11	5.20e-11	1.50e-13

 Table J-13. LCD air emissions (kg/functional unit)

Material	Disposition	Upstream	Mfg	Use	EOL	Total	Fraction of Total
Halogenated hydrocarbons (unspecified)	Air	1.12e-06	1.42e-14	0	-4.93e-16	1.12e-06	3.24e-09
Halogenated matter (unspecified)	Air	4.67e-19	0	0	0	4.67e-19	1.35e-21
HALON-1301	Air	7.95e-08	2.47e-11	0	-8.75e-13	7.95e-08	2.30e-10
HCFC-225ca	Air	0	1.40e-04	0	0	1.40e-04	4.05e-07
HCFC-225cb	Air	0	1.40e-04	0	0	1.40e-04	4.05e-07
Heptane	Air	1.09e-06	7.77e-05	0	0	7.88e-05	2.28e-07
Hexamethyldisilizane	Air	0	1.37e-06	0	0	1.37e-06	3.97e-09
Hexane	Air	2.86e-04	2.86e-05	2.24e-06	-2.73e-06	3.14e-04	9.07e-07
Hydrocarbons, remaining unspeciated	Air	9.30e-03	7.75e-03	0	-6.51e-04	1.64e-02	4.74e-05
Hydrochloric acid	Air	1.50e-03	6.58e-02	4.02e-02	-6.88e-04	1.07e-01	3.09e-04
Hydrofluoric acid	Air	2.27e-04	5.27e-02	5.02e-03	-1.47e-04	5.78e-02	1.67e-04
Hydrogen	Air	2.99e-04	1.33e-04	0	0	4.32e-04	1.25e-06
Hydrogen cyanide	Air	3.83e-06	0	0	0	3.83e-06	1.11e-08
Hydrogen sulfide	Air	1.55e-05	1.51e-04	0	-8.88e-06	1.57e-04	4.55e-07
Indeno(1,2,3-cd)pyrene	Air	3.44e-10	1.57e-09	2.46e-09	-6.23e-11	4.31e-09	1.25e-11
Iodine	Air	2.95e-07	0	0	0	2.95e-07	8.52e-10
Iron	Air	4.76e-05	1.85e-06	0	-1.11e-07	4.94e-05	1.43e-07
Isophorone	Air	4.70e-07	2.53e-06	1.94e-05	-5.67e-07	2.18e-05	6.32e-08
Isopropyl alcohol	Air	0	1.78e-02	0	0	1.78e-02	5.13e-05
Isopropylpropionate	Air	9.91e-11	0	0	0	9.91e-11	2.87e-13
ITO etchant, unspecified	Air	0	6.86e-03	0	0	6.86e-03	1.98e-05
Lanthanum	Air	2.63e-08	0	0	0	2.63e-08	7.61e-11
Lead	Air	3.13e-06	6.63e-07	4.76e-06	4.76e-06	1.33e-05	3.85e-08
Lead (Pb, ore)	Air	1.48e-06	0	0	0	1.48e-06	4.29e-09
Magnesium	Air	4.06e-05	4.80e-05	3.68e-04	-1.07e-05	4.46e-04	1.29e-06
Manganese	Air	2.94e-06	1.24e-06	1.70e-05	-3.23e-06	1.80e-05	5.19e-08
Manganese (Mn, ore)	Air	0	3.67e-06	0	0	3.67e-06	1.06e-08
Mercaptans	Air	1.81e-07	0	0	0	1.81e-07	5.24e-10
Mercury	Air	8.53e-07	4.57e-07	2.80e-06	-8.64e-08	4.03e-06	1.16e-08
Metals, remaining unspeciated	Air	2.64e-05	1.60e-08	0	-7.10e-10	2.64e-05	7.63e-08
Methane	Air	3.54e+00	1.22e-01	2.41e-01	-2.82e-02	3.87e+00	1.12e-02
Methanol	Air	1.11e-06	0	0	0	1.11e-06	3.20e-09
Methyl chloride	Air	4.30e-07	2.31e-06	1.77e-05	-5.18e-07	2.00e-05	5.77e-08
Methyl ethyl ketone	Air	3.16e-07	2.43e-06	1.31e-05	-3.81e-07	1.54e-05	4.46e-08
Methyl ethyl ketone	Treatment	0	1.34e-04	0	0	1.34e-04	3.88e-07
Methyl hydrazine	Air	1.38e-07	7.42e-07	5.69e-06	-1.66e-07	6.40e-06	1.85e-08
Methyl methacrylate	Air	1.62e-08	8.73e-08	6.69e-07	-1.95e-08	7.53e-07	2.18e-09
Methyl tert-butyl ether	Air	2.84e-08	1.52e-07	1.17e-06	-3.81e-07	9.71e-07	2.81e-09
Molybdenum	Air	2.98e-07	6.21e-07	2.18e-07	-1.35e-08	1.12e-06	3.25e-09
Monosilane	Air	0	1.54e-03	0	0	1.54e-03	4.44e-06
Naphthalene	Air	1.30e-07	7.67e-07	6.82e-07	-1.90e-08	1.56e-06	4.51e-09
N-bromoacetamide	Air	0	9.18e-03	0	0	9.18e-03	2.65e-05
Nickel	Air	8.06e-06	5.57e-05	2.56e-05	-2.98e-06	8.64e-05	2.50e-07
Nitric acid	Air	0	2.69e-04	0	0	2.69e-04	7.76e-07
Nitrogen dioxide	Air	3.08e-02	0	0	4.46e-04	3.12e-02	9.03e-05
Nitrogen fluoride	Air	0	2.45e-01	0	0	2.45e-01	7.09e-04
Nitrogen oxides	Air	6.56e-01	7.62e-01	4.39e-01	-1.36e-02	1.84e+00	5.33e-03
Nitrous oxide	Air	1.24e-03	1.20e-03	1.27e-03	-7.19e-04	2.98e-03	8.63e-06

Table J-13. LCD air emissions (kg/functional unit)

Material	Disposition	Upstream	Mfg	Use	EOL	Total	Fraction of Total
Nonmethane hydrocarbons, remaining							
unspeciated	Air	2.07e-01	8.87e-03	0	-1.26e-03	2.15e-01	6.22e-04
n-Propane	Air	4.49e-05	8.67e-08	0	-2.40e-07	4.48e-05	1.29e-07
Other organics	Air	4.45e-01	1.35e-02	0	-2.41e-03	4.56e-01	1.32e-03
o-xylene	Air	4.83e-06	2.24e-07	2.13e-08	-2.50e-08	5.05e-06	1.46e-08
Pentane	Air	4.18e-04	4.10e-05	0	-3.85e-06	4.55e-04	1.31e-06
Perfluoroethane	Air	5.36e-06	0	0	0	5.36e-06	1.55e-08
Perfluoromethane	Air	4.83e-05	0	0	0	4.83e-05	1.40e-07
Phenanthrene	Air	4.92e-09	1.92e-08	9.35e-08	-2.66e-09	1.15e-07	3.33e-10
Phenol	Air	9.34e-07	6.98e-08	5.36e-07	-1.56e-08	1.52e-06	4.41e-09
Phosphine	Air	0	6.26e-02	0	0	6.26e-02	1.81e-04
Phosphoric acid	Air	0	4.85e-05	0	0	4.85e-05	1.40e-07
Phosphorus (yellow or white)	Air	7.54e-07	6.20e-07	1.85e-06	-7.84e-08	3.14e-06	9.09e-09
Phosphorus pentoxide	Air	2.56e-10	0	0	0	2.56e-10	7.40e-13
PM	Air	9.16e-02	6.99e-03	0	-1.24e-02	8.62e-02	2.49e-04
PM-10	Air	3.45e-07	6.85e-03	2.16e-02	3.43e-06	2.84e-02	8.21e-05
Polycyclic aromatic hydrocarbons	Air	6.77e-06	2.84e-12	0	-1.01e-13	6.77e-06	1.96e-08
Polyimide, unspecified	Air	0	1.40e-04	0	0	1.40e-04	4.04e-07
Potassium	Air	1.10e-05	0	0	0	1.10e-05	3.19e-08
Process material for backlight assembly	Air	0	7.03e-05	0	0	7.03e-05	2.03e-07
Propionaldehvde	Air	3.08e-07	1.66e-06	1.27e-05	-3.71e-07	1.43e-05	4.14e-08
Propylene	Air	1.18e-05	0	0	0	1.18e-05	3.42e-08
Pyrene	Air	1.08e-09	4.51e-09	1.24e-08	-3.30e-10	1.77e-08	5.11e-11
Scandium	Air	6.40e-09	0	0	0	6.40e-09	1.85e-11
Selenium	Air	1.21e-06	6.11e-06	4 36e-05	-1 27e-06	4 97e-05	1 44e-07
Silicon	Air	2.89e-04	8 31e-07	0	-4 96e-08	2 90e-04	8 37e-07
Sodium	Air	7.73e-06	4.92e-06	0	-2 93e-07	1 24e-05	3 57e-08
Strontium	Air	1.65e-06	0	0	0	1.65e-06	4 78e-09
Styrene	Air	2.03e-08	1.09e-07	8 37e-07	-2 44e-08	9.42e-07	2 72e-09
Sulfur dioxide	Air	4.63e-02	2.95e-01	9.30e-01	2.44e 00	1.27e+00	3.68e-03
Sulfur beyafluoride	Air	4.050 02	7 30e-03	0.500 01	2.900 04	7.30e-03	2.11e-05
Sulfur oxides	Air	2 57e-02	4.07e-02	0	-1 93e-02	4.71e-02	1 36e-04
Sulfuric acid	Air	6.31e-07	4.070-02	0	-1.950-02	6.31e-07	1.30c-04
Tars (unspecified)	Air	3.000.13	0	0	0	3.00e 13	8.020-09
Tatrachloroothylono	Air	2 400 08	1 880 07	1 440 06	4 140 08	1.620.06	4.6% 00
Tetramathyl ammonium hydroxida	All	3.490-08	6.420.01	1.440-00	-4.140-08	6.420.01	4.086-09
Thellium	Air	4 362 00	0.436-01	0	0	4.36-01	1.806-05
	Alf	4.306-09	0	0	0	4.306-09	1.200-11
Tin	Air	8.21 ₂ .00	0	0	0	1.41e-08	4.096-11
1 III Titenium	Alf	0.216-09	0	0	0	8.21e-09	2.576-11
	Air	2.858-06	674.04	0	1.02.07	2.85e-06	8.246-09
TOCs, remaining unspeciated	Air	1.05.05	6.74e-04	2.15e-03	4.08e-07	2.82e-03	8.16e-06
Toluene	Air	1.05e-05	5.95e-05	9.49e-06	-2.95e-07	7.92e-05	2.29e-07
	Air	0	0	0	5.85e-10	5.85e-10	1.69e-12
Unspecified LCD process material	Air	0	4.49e-02	0	0	4.49e-02	1.30e-04
Uranium	Air	1.39e-08	0	0	0	1.39e-08	4.01e-11
v anadium	Air	1.77e-05	3.22e-05	6.57e-06	-9.16e-07	5.56e-05	1.61e-07
Vinyl acetate	Air	6.16e-09	3.31e-08	2.54e-07	-7.42e-09	2.86e-07	8.28e-10
Vinyl chloride	Air	0	0	0	1.17e-09	1.17e-09	3.38e-12
Waste metals, unspecified	Air	6.33e-06	0	0	0	6.33e-06	1.83e-08

 Table J-13. LCD air emissions (kg/functional unit)

Material	Disposition	Upstream	Mfg	Use	EOL	Total	Fraction of Total		
Xylene (mixed isomers)	Air	9.57e-06	1.49e-07	1.24e-06	-2.71e-08	1.09e-05	3.16e-08		
Zinc (elemental)	Air	6.60e-06	1.80e-05	5.69e-06	-1.81e-07	3.01e-05	8.69e-08		
Zirconium	Air	6.71e-09	0	0	0	6.71e-09	1.94e-11		
Total air emissions		1.12e+02	6.48e+01	1.68e+02	1.30e+00	3.46e+02	1.00e+00		

 Table J-13. LCD air emissions (kg/functional unit)

WASTEWATER STREAM Surface water $8.7p$:100 $2.52e$:103 0.2 $2.41e$:100 $2.53e$:103 8.091% Wastewater stream surface water $0.57p$:102 0.0 $2.57e$:102 0.0 $5.97e$:102 0.0 $2.29e$:48 $1.34e$:164 $1.36e$:163 $1.34e$:164 $1.36e$:164 $1.34e$:164 $1.36e$:164 0.0 0.0 $2.29e$:48 $1.34e$:44 0.0080% $A.6ds$:13 $5.26e$:18 0.0 0.0 $0.375e$:48 $3.14E$:04 0.0080% $A.6ds$:13 $5.26e$:48 $3.14E$:04 0.0001% $A.1e^{-110}$ $5.46e$:48 $3.14E$:04 $A.6ds$:04 0.0 0	Material	Disposition	Unstream	Mfg	Use	EOL	Total	% or Fraction of Total
Wasewater stream surface water $8.57e+00$ $2.52e+03$ 0 $2.41e+00$ $2.53e+03$ 8.919 /s Total watevater stream $K 57e-00$ $1.2e+05$ 0 $2.41e+00$ $3.13e+03$ 100.00% WATER POLLUTANTS $1.2e+05$ 0 0 0 $1.74e+11$ $1.34e+04$ $1.76e+08$ 0 0 0 $1.74e+11$ $1.036+11$ Acids (I1) surface water $2.56e+08$ $1.82e+11$ 0 $6.4e+13$ $5.26e+08$ $3.14e+04$ Achohols surface water $5.26e+08$ $1.82e+11$ 0 $6.4e+13$ $5.32e+08$ $3.014-08$ Alkancis surface water $5.26e+08$ $1.32e+06$ 0 <	WASTEWATER STREAM			8			- • • • •	- • • • • • •
Wastewater stream treatment 0 5.97e+02 0 0 5.97e+02 1.9203 0.00.00% Total vastewater stream 8.97e+00 3.12e+03 0.0 2.41e+00 3.12e+03 100.00% MCTER POLLUTANTS 1.1,1-Fichilorochtane surface water 0 0 0 1.74e+11 1.74e+11 1.03E+11 Acids (H+) surface water 1.34e-04 1.76e-08 0 -3.15e-09 1.34e-04 0.0000% Alcohols surface water 3.75e-08 0 0 0 3.75e-08 2.21E-08 Alcohols surface water 1.92e-06 0 0 0 1.92e-06 0.0000% Alkane (suppectified) surface water 1.97e-07 0 0 0 1.97e-07 0.0001% Aluminum (+3) surface water 6.32e-04 0 -2.41e-07 0.0001% Aluminum (+3) surface water 6.32e-04 0 -3.91e-06 6.19e-04 0.0005% Auminum (+3) surface water 2	Wastewater stream	surface water	8 57e+00	2.52e+03	0	-2.41e+00	2.53e+03	80 91%
Tatal wastewater stream 8.57e+00 3.12e+03 0 -2.41e+00 3.12e+03 100.00% WATER POLLUTANTS	Wastewater stream	treatment	0	5.97e+02	0	0	5.97e+02	19.09%
WATER POLLUTANTS 11,17-Fieldrocethane surface water 2,78-14 2,29-08 0 0 2,29-08 1,36-10 1,1,17-Holdrocethane surface water 0 0 0 1,74-11 1,77-17 1,76-11	Total wastewater stream	IS	8.57e+00	3.12e+03	0	-2.41e+00	3.13e+03	100.00%
1,1-Trichloroethane surface water 2,78e-14 2,29e-08 0 0 1,74e-11 1,03E-11 1,2-Dichloroethane surface water 1,34e-04 1,76e-08 0 1,74e-11 1,03E-14 Adschable organic halides surface water 1,34e-04 0,0080% 0 0 2,75e-08 0,315e-09 1,34e-04 0,0080% Adschable surface water 0,35e-08 0 0 0 0 3,75e-08 0,224E-08 Alcabals surface water 5,36e-06 0 0 0 5,36e-08 0,302F-10 Alkenes surface water 1,77e-07 0 0 0 1,77e-07 0.0 0 1,77e-04 0.0366% Alminum (r1) surface water 6,36e-01 0 0 3,31e-03 0 4,36e-10 0,0366% Annonia surface water 6,21e-04 0 0 3,31e-06 6,07e-04 0,0369% Annonia surface water 2,1e-06 0.0 0 1,14e-07 6,00e-04 0,00078% Annonia surface water 8	WATER POLLUTANTS	-						
1.2-Dichloroethane surface water 0 0 1.74e-11 1.74e-11 1.03E-11 Acids (H+) surface water 1.34e-04 1.76e-08 0 0.315e-09 314e-04 0.0008% Acisorbabi corganic halides surface water 3.75e-08 0 0 0 0.375e-08 2.23E-08 Alcohobis surface water 5.75e-06 0 0 0 1.92e-06 0.0001% Alkanc (unspecified) surface water 1.92e-06 0 0 0 1.92e-06 0.0001% Alkanc (unspecified) surface water 6.00e-04 6.88e-06 0 2.44e-07 6.07e-04 0.0361% Aluminum hydroxide surface water 6.23e-04 0 0 -0 4.36e-10 2.60e-10 Anmonia surface water 2.21e-03 0.33e-03 -1.34e-03 0.9732% Animony surface water 2.21e-04 0 0 1.14e-07 6.80E-08 Aromatic hydrocarbons surface water 2.13e-05 1.33e-04 0.0001% Assencio 0.24e-05 0.0001% Bar	1.1.1-Trichloroethane	surface water	2.78e-14	2.29e-08	0	0	2.29e-08	1.36E-08
Acids (H+) surface water 1.34e-04 1.76e-08 0 -3.15e-09 1.34e-04 0.0080% Adsorbable organic hulides surface water 5.26e-08 1.52e-11 0 -6.44e-13 5.26e-08 3.14E-08 Adsorbable surface water 5.75e-08 0 0 0 5.75e-08 2.23E-08 Aldehydes surface water 1.92e-06 0 0 0 1.92e-06 0.0001% Alkance (unspecified) surface water 1.92e-06 0 0 0 1.92e-06 0.0001% Alkance (unspecified) surface water 4.36e-10 0 0 0 4.36e-10 2.06e-10 2.06e-10 2.06e-10 2.06e-10 2.06e-10 2.06e-10 2.06e-10 2.06e-10 2.07e-01 0 1.14e-07 6.80E-08 A.36e-05 1.31e-03 0 1.34e-06 0.0007% Arminio ins surface water 0 1.14e-07 6.80E-08 Arsenio 1.14e-07 6.80E-08 0 1.14e-07 6.80E-08 0.0001%	1.2-Dichloroethane	surface water	0	0	0	1.74e-11	1.74e-11	1.03E-11
Adsorbable organic halides surface water $5.26e-08$ $1.82e-11$ 0 $-6.44e-13$ $5.26e-08$ $2.21E-06$ Alcohols surface water $5.76e-08$ 0 0 0 $3.75e-08$ $2.22E-06$ Alcohols surface water $1.92e-06$ 0 0 0 $5.9e-10$ $3.03E-10$ Alkenes surface water $1.92e-06$ 0 0 0 1.77e-07 $1.06E-07$ Aluminum lydroxide surface water $4.36e-10$ 0 0 $-2.44e-07$ $6.07e-04$ 0.0399% Annonia surface water $2.3e-05$ $1.33e-03$ 0 $-4.38e-05$ $1.31e-03$ 0.0782% Annonia surface water $2.3e-06$ 0 0 0 $1.14e-07$ $6.80E-08$ Aromatic hydrocarbons surface water $1.24e-06$ 0 0 0 $1.44e-07$ $6.80E-08$ Aromatic hydrocarbons surface water $1.36e-04$ 0 0 0 0 0 $0.14e-07$ $0.80E-08$ Barium myds surface water $1.$	Acids (H+)	surface water	1.34e-04	1.76e-08	0	-3.15e-09	1.34e-04	0.0080%
Alcehols surface water $3.75e-08$ 0 0 0 $3.75e-08$ 223E-08 Aldelydes surface water $5.08e-10$ 0 0 0 5.08e-10 3.03E-10 Alken (unspecified) surface water $1.72e-07$ 0 0 0 1.72e-07 1.06F-07 Aluminum hydroxide surface water $6.00e-04$ $6.88e-06$ 0 $-2.44e-07$ $6.07e-04$ 0.06F-07 Aluminum hydroxide surface water $6.23e-04$ 0 0 $4.38e-10$ 2.06F-10 Annonia surface water $6.23e-04$ 0 0 $-3.91e-06$ $6.19e-04$ 0.0782% Annonia ions surface water $2.51e-05$ $1.33e-03$ 0 $4.38e-06$ 0.00782% Arninonj surface water $0.114e-07$ 0 0 $1.14e-07$ 6.80e-08 Arninic hydrocarbons surface water $1.24e-06$ 0 $0.534e-09$ $1.24e-06$ 0.0001% Barium enpds surface water $1.86e-04$ 0 0 $0.74e-02$ 0 $0.574e-02$ 0.0001%	Adsorbable organic halides	surface water	5.26e-08	1.82e-11	0	-6.44e-13	5.26e-08	3.14E-08
Aldehydes surface water 5.08e-10 0 0 5.08e-10 3.03E-10 Alkene (unspecified) surface water 1.92e-06 0 0 1.92e-06 0.0001% Alkenes surface water 1.72e-77 0 0 0 1.72e-07 1.06E-07 Aluminum (+3) surface water 6.00e-04 6.88e-06 0 -2.44e-07 6.07e-04 0.0369% Aluminum hydroxide surface water 2.51e-05 1.33e-03 0 4.38e-05 1.31e-03 0.0782% Annonis ions surface water 2.51e-05 1.32e-03 0 1.14e-07 6.0005% Aromatic hydrocarbons surface water 0 1.14e-07 0 0 1.4e-07 6.0005% Arsenic cmpds surface water 1.86e-04 0 0 0 1.8ee-04 0.0001% Barium empds surface water 1.86e-04 0 0 0 5.34e-02 1.8e764 0.0001% Borbon treatment 0 5.74e	Alcohols	surface water	3.75e-08	0	0	0	3.75e-08	2.23E-08
Alkane (unspecified) surface water $1.92e-06$ 0 0 0 $1.92e-06$ 0.0001% Alkenes surface water $1.77e-07$ 0 0 0 $1.77e-07$ $1.06E-07$ Aluminum hydroxide surface water $4.36e-10$ $0.43e-10$ $0.391e-06$ $6.9e-04$ 0.389% Anmonia surface water $6.23e+04$ 0 $0.391e-06$ $6.19e-04$ 0.089% Anmonia surface water $0.114e-07$ 0 $0.114e-07$ $6.80e-08$ Aromatic hydrocarbons surface water $0.114e-07$ 0.0 $0.114e-07$ $6.80e-08$ Arsenic surface water $1.24e-06$ 0.0 $0.114e-07$ 0.0 $0.114e-07$ $6.80e-08$ Arsenic surface water $1.24e-06$ 0.0 0.0 $1.34e-06$ 0.0001% Barium sulfac surface water $1.98e-04$ 0.0 $0.84e-04$ 0.0 $0.84e-04$ 0.0 $0.84e-04$ 0.0 $0.84e-04$ 0.0001%	Aldehydes	surface water	5.08e-10	0	0	0	5.08e-10	3.03E-10
Alkenes surface water 1.77e-07 0 0 0 1.77e-07 1.06E-07 Aluminum (43) surface water 6.00e-04 6.88e-06 0 -2.44e-07 6.07e-04 0.0361% Aluminum hydroxide surface water 6.23e-04 0 0 -3.91e-06 6.19e-04 0.0369% Anmonia surface water 2.51e-05 1.33e-03 0 -4.38e-05 1.31e-03 0.0782% Antimiony surface water 2.51e-05 1.33e-03 0 -1.51e-10 8.46e-06 0.0005% Aromatic hydrocarbons surface water 8.45e-06 4.25e-09 0 -1.14e-07 6.80E-08 Arsenic empds surface water 1.24e-06 0 0 1.38e-04 0.0001% Barium cmpds surface water 1.86e-04 0 0 0 1.38e-04 0.0011% Borace surface water 1.93e-06 0 0 1.31e-06 0.0001% Bora surface water 1.97e-02 0	Alkane (unspecified)	surface water	1.92e-06	0	0	0	1.92e-06	0.0001%
Aluminum (+3) surface water 6.00e-04 6.88e-06 0 2.44e-07 6.07e-04 0.0361% Aluminum hydroxide surface water 4.36e-10 0 0 0 4.36e-10 2.00E-10 Ammonia surface water 2.51e-05 1.33e-03 0 4.38e-05 1.31e-03 0.0782% Antimony surface water 0 1.14e-07 0 0 1.14e-07 6.80E-08 Aromatic hydrocarbons surface water 0 1.14e-07 0 0 1.14e-07 6.80E-08 Arsenic surface water 1.24e-06 0 0 1.34e-06 0.0001% Barium cmpds surface water 1.86e-04 0 0 0 1.86e-04 0.0111% Benzene surface water 1.93e-06 0 1.31e-06 0.0001% BOD treatment 0 5.74e-02 0 0 5.74e-02 3.4201% Bora caid surface water 0 4.58e-06 0 0 <	Alkenes	surface water	1.77e-07	0	0	0	1.77e-07	1.06E-07
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Aluminum (+3)	surface water	6.00e-04	6.88e-06	0	-2.44e-07	6.07e-04	0.0361%
Armonia surface water 6.23e-04 0 0 3.91e-06 6.19e-04 0.0369% Ammonia ions surface water 0 1.14e-07 0 0 1.14e-07 6.80E-08 Animony surface water 0 1.14e-07 0 0 1.14e-07 6.80E-08 Aromatic hydrocarbons surface water 0 1.14e-07 0 0 1.14e-07 6.80E-08 Arsenic cmpds surface water 1.24e-06 0 0 5.34e-09 1.24e-06 0.0001% Barium cmpds surface water 1.86e-04 0 0 0 1.86e-04 0.0111% Benzene surface water 1.99e-06 0 0 1.74e-11 1.93e-06 0.0001% BOD surface water 7.72e-04 2.79e-02 0 0 5.76e-07 3.4201% Boric acid surface water 5.76e-07 0 0 0 5.78e-07 1.32E-06 0.0001% Boric acid surface water <	Aluminum hydroxide	surface water	4.36e-10	0	0	0	4.36e-10	2.60E-10
Ammonia ions surface water $2.51e-05$ $1.33e-03$ 0 $-4.38e-05$ $1.31e-03$ 0.0782% Antimony surface water 0 $1.14e-07$ 0 0 $1.14e-07$ $6.801e-08$ Aromatic hydrocarbons surface water 0 $1.14e-07$ 0 0 $1.14e-07$ $6.801e-08$ Arsenic surface water $1.24e-06$ 0 0 $5.34e-09$ $1.24e-06$ 0.0001% Barium cmpds surface water $1.86e-04$ 0.0111% 0 0 $1.86e-04$ 0.0001% Borne surface water $1.93e-06$ 0 0 $1.8e-04$ 0.0001% BOD surface water $1.72e-04$ $2.79e-02$ 0 0 $5.76e-07$ $3.221e-07$	Ammonia	surface water	6.23e-04	0	0	-3.91e-06	6.19e-04	0.0369%
Antimory surface water 0 1.14e-07 0 0 1.14e-07 6.80E-08 Aromatic hydrocarbons surface water 0 1.14e-07 0 0 1.14e-07 6.80E-08 Arsenic mpds surface water 1.24e-06 0 0 5.34e-06 0.0001% Barium cmpds surface water 1.28e-04 0 0 0 1.24e-06 0.0001% Barium sulfate surface water 1.86e-04 0 0 0 1.86e-04 0.0001% Benzene surface water 7.72e-04 2.79e-02 0 -3.18e-04 2.83e-02 1.6873% BOD treatment 0 5.74e-02 0 0 5.74e-02 3.4201% Borax treatment 0 1.31e-06 0 0 5.74e-07 3.32E-07 Boron surface water 9.79e-08 0 0 0 1.14e-07 6.80E-08 Cadmium surface water 1.31e-03 0 0 1.31e	Ammonia ions	surface water	2.51e-05	1.33e-03	0	-4.38e-05	1.31e-03	0.0782%
Aromatic hydrocarbons surface water 8.45c-06 4.25c-09 0 -1.51c-10 8.46c-06 0.0005% Arsenic surface water 0 1.14c-07 0 0 1.14c-07 6.80E-08 Arsenic cmpds surface water 1.24c-06 0 0 5.34c-09 1.24c-06 0.0001% Barium empds surface water 1.36c-04 0 0 1.86c-04 0.0011% Benzne surface water 1.93c-06 0 0 1.74c-11 1.93c-06 0.0001% BOD surface water 7.72c-04 2.79e-02 0 5.74c-02 0.458c-06 0.0001% BOD treatment 0 1.31c-06 0 0 1.31c-06 0.0001% Borax treatment 0 1.31c-07 0 0 9.79c-08 0.00003% Boron (B1I) surface water 9.79c-08 0 0 9.79c-08 0.83E-08 Cadmium surface water 1.89c-07 1.42c-11 0	Antimony	surface water	0	1.14e-07	0	0	1.14e-07	6.80E-08
Arsenicsurface water01.14e-07001.14e-076.80E-08Arsenic cmpdssurface water1.24e-06005.34e-091.24e-060.0011%Barium suffatesurface water1.36e-040001.86e-040.0011%Benzenesurface water1.36e-04001.74e-111.93e-060.0001%BODsurface water7.72e-042.79e-020-3.18e-042.83e-021.6873%BODtreatment05.74e-02005.74e-023.4201%Boraxtreatment01.31e-06001.31e-060.0001%Boraxsurface water5.57e-070005.57e-073.32E-07Boronsurface water9.79e-08009.79e-085.83E-08Cadmiumsurface water01.14e-07001.14e-076.80E-08Cadmiumsurface water1.37e-030001.37e-030.0816%Cadmium (+2)surface water1.32e-01001.74e-111.03E-11Carbon ternchloridesurface water5.29e-090001.37e-030.0816%Carbon ternchloridesurface water5.28e-060005.32e-093.35E-09Chlorinesurface water5.28e-070005.35E-093.35E-09Choride ionssurface water5.28e-07000<	Aromatic hydrocarbons	surface water	8.45e-06	4.25e-09	0	-1.51e-10	8.46e-06	0.0005%
Arsenic empdssurface water1.24e-06005.34e-091.24e-060.0001%Barium empdssurface water8.36e-051.36e-0807.51e-088.37e-050.0050%Barium sulfatesurface water1.86e-040001.86e-040.0011%Benzenesurface water1.93e-0601.74e-111.93e-060.0001%BODsurface water7.72e-042.79e-0203.18e-042.83e-021.6873%BODtreatment05.74e-02005.74e-023.4201%Boraxtreatment01.31e-060.0001%0.0001%Boric acidsurface water5.57e-0700005.74e-02Boronsurface water9.79e-080004.58e-060.0001%Boron (B III)surface water9.79e-080001.14e-076.80E-08Cadinium empdssurface water1.89e-071.14e-07001.14e-076.80E-08Calcium (+2)surface water1.37e-030001.37e-030.0816%Carbon tetrachloridesurface water2.99e-030002.69e-030.1610%Cesium (+2)surface water5.92e-090005.92e-093.53E-09Chorinesurface water2.33e-013.12e-010-1.58e-025.29e-013.15275%Chloride ionssurface water6.36e-07	Arsenic	surface water	0	1.14e-07	0	0	1.14e-07	6.80E-08
Barium empds surface water 8.36e-05 1.36e-08 0 7.51e-08 8.37e-05 0.0050% Barium sulfate surface water 1.93e-06 0 0 0 1.86e-04 0.0011% Benzene surface water 1.93e-06 0 0 1.74e-11 1.93e-06 0.0001% BOD surface water 7.72e-04 2.79e-02 0 0 5.74e-02 3.48e-04 2.83e-02 3.4201% Borax treatment 0 1.31e-06 0 0 1.31e-06 0.0001 5.74e-02 3.4201% Borax treatment 0 1.31e-06 0 0 5.74e-02 3.4201% Boran surface water 9.79e-08 0 0 0 4.58e-06 0.0001% Boron (B III) surface water 1.89e-07 1.14e-07 6.80E-08 Cadmium empds surface water 1.89e-07 1.13E-07 Caldmium empds surface water 1.26e-03 0 0 0 1.37e-03	Arsenic cmpds	surface water	1.24e-06	0	0	5.34e-09	1.24e-06	0.0001%
Barium sulfate surface water 1.86e-04 0 0 0 1.86e-04 0.0111% Benzene surface water 1.93e-06 0 0 1.74e-11 1.93e-06 0.0001% BOD surface water 7.72e-04 2.79e-02 0 -3.18e-04 2.83e-02 1.6873% BOD treatment 0 5.74e-02 0 0 5.74e-02 3.4201% Borax treatment 0 1.31e-06 0.0001% 5.57e-07 3.32E-07 Boron surface water 9.79e-08 0 0 9.79e-08 0.0001% Cadmium surface water 0 1.42e-07 0 0 1.13e-07 Cadmium enpds surface water 1.89e-07 1.42e-11 0 2.85e-10 1.89e-07 1.13E-07 Calcium (+2) surface water 1.37e-03 0 0 0 1.37e-03 0.0816% Carbonate ion surface water 2.69e-03 0 0 0 5.92e-09	Barium cmpds	surface water	8.36e-05	1.36e-08	0	7.51e-08	8.37e-05	0.0050%
Benzene surface water 1.93e-06 0 1.74c-11 1.93e-06 0.0001% BOD surface water 7.72e-04 2.79e-02 0 -3.18e-04 2.83e-02 1.6873% BOD treatment 0 5.74e-02 0 0 5.74e-02 3.4201% Borax treatment 0 1.31e-06 0 0 1.31e-06 0.0001% Boric acid surface water 5.57e-07 0 0 0 5.57e-07 3.32E-07 Boron surface water 9.79e-08 0 0 4.58e-06 0.0001% Boron (B III) surface water 9.79e-08 0 0 1.14e-07 6.80E-08 Cadmium engds surface water 1.89e-07 1.14e-07 0 0 1.37e-03 0.0816% Carbon tetrachloride surface water 2.69e-03 0 0 1.37e-03 0.0816% Carbon tetrachloride surface water 5.92e-09 0 0 1.57e-07 3.32E-07 </td <td>Barium sulfate</td> <td>surface water</td> <td>1.86e-04</td> <td>0</td> <td>0</td> <td>0</td> <td>1.86e-04</td> <td>0.0111%</td>	Barium sulfate	surface water	1.86e-04	0	0	0	1.86e-04	0.0111%
BOD surface water 7.72e-04 2.79e-02 0 -3.18e-04 2.83e-02 1.6873% BOD treatment 0 5.74e-02 0 0 5.74e-02 3.4201% Borax treatment 0 1.31e-06 0 0 1.31e-06 0.0001% Bora caid surface water 5.57e-07 0 0 0 5.57e-07 3.32E-07 Boron surface water 9.79e-08 0 0 0 4.58e-06 0.0003% Boron (B III) surface water 9.79e-08 0 0 0 9.79e-08 5.83E-08 Cadmium empds surface water 1.89e-07 1.42e-11 0 2.85e-10 1.89e-07 1.13E-07 Calcium (+2) surface water 1.37e-03 0 0 0 1.37e-03 0.0816% Carbon tetrachloride surface water 2.69e-03 0 0 0 2.69e-03 0.1601% Carbon tetrachloride surface water 5.92e-09 0 </td <td>Benzene</td> <td>surface water</td> <td>1.93e-06</td> <td>0</td> <td>0</td> <td>1.74e-11</td> <td>1.93e-06</td> <td>0.0001%</td>	Benzene	surface water	1.93e-06	0	0	1.74e-11	1.93e-06	0.0001%
BODtreatment0 $5.74e-02$ 00 $5.74e-02$ 3.4201% Boraxtreatment0 $1.31e-06$ 00 $1.31e-06$ 0.0001% Boric acidsurface water $5.57e-07$ 000 $5.57e-07$ $3.32E-07$ Boronsurface water0 $4.58e-06$ 00 $4.58e-06$ 0.0001% Boron (B III)surface water9.79e-08000 $9.79e-08$ $5.83E-08$ Cadmium cmpdssurface water1.89e-07 $1.14e-07$ 00 $1.14e-07$ $6.80E-08$ Cadrium cmpdssurface water $1.37e-03$ 000 $1.37e-03$ 0.0816% Carbon tetrachloridesurface water $2.69e-03$ 000 $2.69e-03$ 0.0816% Carbon tetrachloridesurface water $2.69e-03$ 000 $2.69e-03$ 0.1610% Cesium (+2)surface water $2.33e-01$ $3.12e-01$ 0 $1.58e-02$ $5.29e-01$ 3.5275% Chloride ionssurface water $2.33e-01$ $3.12e-01$ 0 0 $6.36e-07$ $3.79E-07$ Chloroformsurface water $6.23e-06$ $3.49e-09$ 0 0 $0.84e-06$ 0.0005% Chromium (III)surface water $7.67e-03$ $8.20e-02$ 0 $1.80e-09$ $6.23e-06$ 0.0005% Chromium (VI)surface water $7.67e-03$ $8.20e-02$ 0 0 0 $0.87e-02$ 5.1827% CODtraf	BOD	surface water	7.72e-04	2.79e-02	0	-3.18e-04	2.83e-02	1.6873%
Borax treatment 0 1.31e-06 0 0 1.31e-06 0.0001% Boria acid surface water 5.57e-07 0 0 0 5.57e-07 3.32E-07 Boron surface water 0 4.58e-06 0 0 4.58e-06 0.0003% Boron (B III) surface water 9.79e-08 0 0 9.79e-08 5.83E-08 Cadmium surface water 1.89e-07 1.42e-01 0 0.114e-07 6.80E-08 Cadmium empds surface water 1.89e-07 1.42e-11 0 2.85e-10 1.89e-07 1.015E-01 Carbon tetrachloride surface water 1.37e-03 0 0 0 1.37e-03 0.0816% Carbon tetrachloride surface water 2.69e-03 0 0 2.69e-03 0.0610% Carbon tetrachloride surface water 2.32e-09 0 0 0 5.29e-01 3.52F9% Chloride ions surface water 2.33e-01 3.12e-01 0	BOD	treatment	0	5.74e-02	0	0	5.74e-02	3.4201%
Boric acid surface water 5.57e-07 0 0 5.57e-07 3.32E-07 Boron surface water 0 4.58e-06 0 0 4.58e-06 0.0003% Boron (B III) surface water 9.79e-08 0 0 9.79e-08 5.83E-08 Cadmium surface water 0 1.14e-07 0 0 1.14e-07 6.80E-08 Cadmium empds surface water 1.89e-07 1.42e-11 0 2.85e-10 1.89e-07 1.13E-07 Calcium (+2) surface water 1.37e-03 0 0 0 1.37e-03 0.0816% Carbon tetrachloride surface water 2.69e-03 0 0 2.69e-03 0.1601% Carbon tetrachloride surface water 2.69e-03 0 0 2.59e-01 31.5275% Chlorine surface water 2.33e-01 3.12e-01 0 -1.58e-02 5.29e-01 31.5275% Chlorine surface water 2.36e-07 0 0 8.84e-06	Borax	treatment	0	1.31e-06	0	0	1.31e-06	0.0001%
Boron surface water 0 4.58e-06 0 4.58e-06 0.0003% Boron (B III) surface water 9.79e-08 0 0 9.79e-08 5.83E-08 Cadmium surface water 0 1.14e-07 0 0 1.14e-07 6.80E-08 Cadmium empds surface water 1.89e-07 1.42e-11 0 2.85e-10 1.89e-07 1.13E-07 Calcium (+2) surface water 0 0 0 1.74e-11 1.74e-11 1.03E-11 Carbon tetrachloride surface water 2.69e-03 0 0 0 2.69e-03 0.160 Cesium (+2) surface water 5.92e-09 0 0 5.92e-09 3.53E-09 Chloride ions surface water 2.33e-01 3.12e-01 0 -1.58e-02 5.29e-01 31.5275% Chloroform surface water 2.87e-12 0 0 1.42e-11 1.21E-11 Chromium (III) surface water 6.23e-06 3.49e-09 1.80e-09 5.	Boric acid	surface water	5.57e-07	0	0	0	5.57e-07	3.32E-07
Boron (B III) surface water 9.79e-08 0 0 9.79e-08 5.83E-08 Cadmium surface water 0 1.14e-07 0 0 1.14e-07 6.80E-08 Cadmium empds surface water 1.89e-07 1.42e-11 0 2.85e-10 1.89e-07 1.13E-07 Calcium (+2) surface water 1.37e-03 0 0 0 1.37e-03 0.0816% Carbon tetrachloride surface water 2.69e-03 0 0 1.37e-03 0.0816% Carbon tetrachloride surface water 2.69e-03 0 0 2.69e-03 0.0816% Carbon tetrachloride surface water 2.69e-03 0 0 0 2.69e-03 0.1601% Cesium (+2) surface water 2.32e-09 0 0 0 5.92e-03 0.1601% Chloride ions surface water 2.33e-01 3.12e-01 0 1.15e-02 5.29e-01 3.152r5% Chloroforn surface water 2.87e-12 0 <	Boron	surface water	0	4.58e-06	0	0	4.58e-06	0.0003%
Cadmium surface water 0 1.14e-07 0 0 1.14e-07 6.80E-08 Cadmium empds surface water 1.89e-07 1.42e-11 0 2.85e-10 1.89e-07 1.13E-07 Calcium (+2) surface water 1.37e-03 0 0 0 1.37e-03 0.0816% Carbon tetrachloride surface water 0 0 0 1.74e-11 1.03E-11 Carbon tetrachloride surface water 2.69e-03 0 0 0.269e-03 0.1601% Cesium (+2) surface water 2.39e-09 0 0 -1.58e-02 5.29e-01 31.5275% Chlorine surface water 2.33e-01 3.12e-01 0 -1.74e-11 2.02e-11 1.21E-11 Chronine surface water 2.87e-12 0 0 1.74e-07 3.79E-07 Chloroform surface water 6.23e-06 3.49e-09 0 1.80e-09 6.23e-06 0.0004% Chromium (III) surface water 7.70e-10 0	Boron (B III)	surface water	9.79e-08	0	0	0	9.79e-08	5.83E-08
Cadmium empdssurface water1.89e-071.42e-1102.85e-101.89e-071.13E-07Calcium (+2)surface water1.37e-030001.37e-030.0816%Carbon tetrachloridesurface water0001.74e-111.74e-111.03E-11Carbon tetrachloridesurface water2.69e-030002.69e-030.1601%Carbon tetrachloridesurface water5.92e-090002.69e-030.1601%Cesium (+2)surface water5.92e-090005.92e-093.53E-09Chloride ionssurface water2.33e-013.12e-010-1.58e-025.29e-0131.5275%Chlorinesurface water2.87e-12001.74e-112.02e-111.21E-11Chromiumsurface water08.84e-06008.84e-060.0005%Chromium (III)surface water6.23e-063.49e-0901.80e-096.23e-060.0004%Chromium (VI)surface water7.70e-100007.70e-104.59E-10CODsurface water7.67e-038.20e-020-2.69e-038.70e-022.325%CODtreatment03.90e-02003.89e-030.2320%Colon bacillus (bacteria in largesurface water03.89e-03003.89e-030.2320%Coppersurface water03.80e-062.84e-100-1	Cadmium	surface water	0	1.14e-07	0	0	1.14e-07	6.80E-08
Calcium (+2) surface water 1.37e-03 0 0 1.37e-03 0.0816% Carbon tetrachloride surface water 0 0 0 1.37e-03 0.0816% Carbon tetrachloride surface water 2.69e-03 0 0 1.74e-11 1.03E-11 Carbonate ion surface water 2.69e-03 0 0 0 2.69e-03 0.1601% Cesium (+2) surface water 5.92e-09 0 0 0 5.92e-09 3.53E-09 Chloride ions surface water 6.36e-07 0 0 -1.58e-02 5.29e-01 31.5275% Chloride ions surface water 2.87e-12 0 0 1.74e-11 2.02e-11 1.21E-11 Chromium surface water 0 8.84e-06 0 0 8.84e-06 0.0005% Chromium (III) surface water 2.33e-07 0 1.80e-09 5.29e-07 3.15E-07 Cobalt (Co I, Co II, Co III) surface water 7.70e-10 0 0	Cadmium cmpds	surface water	1.89e-07	1.42e-11	0	2.85e-10	1.89e-07	1.13E-07
Carbon tetrachloride surface water 0 0 1.74e-11 1.74e-11 1.03E-11 Carbonate ion surface water 2.69e-03 0 0 0 2.69e-03 0.1601% Cesium (+2) surface water 5.92e-09 0 0 0 5.92e-09 3.53E-09 Chloride ions surface water 2.33e-01 3.12e-01 0 -1.58e-02 5.29e-01 31.5275% Chlorine surface water 6.36e-07 0 0 0 6.36e-07 3.79E-07 Chlorine surface water 2.87e-12 0 0 1.74e-11 2.02e-11 1.21E-11 Chromium surface water 6.23e-06 3.49e-09 0 1.80e-09 6.23e-06 0.0004% Chromium (VI) surface water 7.70e-10 0 0 0 7.70e-10 4.59E-10 COD surface water 7.67e-03 8.20e-02 0 2.87e-03 8.70e-02 2.3251% Colon bacillus (bacteria in large intestine)	Calcium (+2)	surface water	1.37e-03	0	0	0	1.37e-03	0.0816%
Carbonate ionsurface water2.69e-03002.69e-030.1601%Cesium (+2)surface water5.92e-090005.92e-093.53E-09Chloride ionssurface water2.33e-013.12e-010-1.58e-025.29e-0131.5275%Chlorinesurface water6.36e-070006.36e-073.79E-07Chloroformsurface water2.87e-12001.74e-112.02e-111.21E-11Chromiumsurface water08.84e-06008.84e-060.0005%Chromium (III)surface water6.23e-063.49e-0901.80e-096.23e-060.0004%Chromium (VI)surface water7.70e-100007.70e-104.59E-10CODsurface water7.67e-038.20e-020-2.69e-038.70e-025.1827%CODtreatment03.90e-02003.90e-022.3251%Colon bacillus (bacteria in large intestine)surface water03.89e-03003.89e-030.2320%Copper (+1 & +2)surface water03.89e-03009.18e-070.0001%	Carbon tetrachloride	surface water	0	0	0	1.74e-11	1.74e-11	1.03E-11
Cesium (+2)surface water5.92e-090005.92e-093.53E-09Chloride ionssurface water2.33e-013.12e-010-1.58e-025.29e-0131.5275%Chlorinesurface water6.36e-070006.36e-073.79E-07Chloroformsurface water2.87e-12001.74e-112.02e-111.21E-11Chromiumsurface water08.84e-06008.84e-060.0005%Chromium (III)surface water6.23e-063.49e-0901.80e-096.23e-060.0004%Chromium (VI)surface water2.94e-072.33e-0701.80e-095.29e-073.15E-07Cobalt (Co I, Co II, Co III)surface water7.70e-100007.70e-104.59E-10CODtreatment03.90e-02003.90e-025.1827%CODtreatment03.90e-02003.90e-022.3251%Colon bacillus (bacteria in large intestine)surface water03.89e-03003.89e-030.2320%Coppersurface water09.18e-07009.18e-070.0001%Copper (+1 & +2)surface water3.80e-062.84e-100-1.01e-113.80e-060.0002%	Carbonate ion	surface water	2.69e-03	0	0	0	2.69e-03	0.1601%
Chloride ionssurface water2.33e-013.12e-010-1.58e-025.29e-0131.5275%Chlorinesurface water6.36e-070006.36e-073.79E-07Chloroformsurface water2.87e-12001.74e-112.02e-111.21E-11Chromiumsurface water08.84e-06008.84e-060.0005%Chromium (III)surface water6.23e-063.49e-0901.80e-096.23e-060.0004%Chromium (VI)surface water2.94e-072.33e-0701.80e-095.29e-073.15E-07Cobalt (Co I, Co II, Co III)surface water7.70e-100007.70e-104.59E-10CODsurface water7.67e-038.20e-020-2.69e-038.70e-025.1827%CODtreatment03.90e-02003.89e-030.2320%Colon bacillus (bacteria in large intestine)surface water03.89e-03009.18e-070.0001%Coppersurface water09.18e-07009.18e-070.0001%	Cesium (+2)	surface water	5.92e-09	0	0	0	5.92e-09	3.53E-09
Chlorinesurface water6.36e-070006.36e-073.79E-07Chloroformsurface water2.87e-12001.74e-112.02e-111.21E-11Chromiumsurface water08.84e-06008.84e-060.0005%Chromium (III)surface water6.23e-063.49e-0901.80e-096.23e-060.0004%Chromium (VI)surface water2.94e-072.33e-0701.80e-095.29e-073.15E-07Cobalt (Co I, Co II, Co III)surface water7.70e-100007.70e-104.59E-10CODtreatment03.90e-02003.90e-025.1827%CODtreatment03.90e-02003.90e-022.3251%Colon bacillus (bacteria in large intestine)surface water09.18e-07009.18e-070.0001%Coppersurface water09.18e-07009.18e-070.0001%Copper (+1 & +2)surface water3.80e-062.84e-100-1.01e-113.80e-060.0002%	Chloride ions	surface water	2.33e-01	3.12e-01	0	-1.58e-02	5.29e-01	31.5275%
Chloroform surface water 2.87e-12 0 0 1.74e-11 2.02e-11 1.21E-11 Chromium surface water 0 8.84e-06 0 0 8.84e-06 0.0005% Chromium (III) surface water 6.23e-06 3.49e-09 0 1.80e-09 6.23e-06 0.0004% Chromium (VI) surface water 2.94e-07 2.33e-07 0 1.80e-09 5.29e-07 3.15E-07 Cobalt (Co I, Co II, Co III) surface water 7.70e-10 0 0 0 7.70e-10 4.59E-10 COD surface water 7.67e-03 8.20e-02 0 -2.69e-03 8.70e-02 5.1827% COD treatment 0 3.90e-02 0 0 3.90e-02 2.3251% Colon bacillus (bacteria in large intestine) surface water 0 3.89e-03 0 0 3.89e-03 0.2320% Copper surface water 0 9.18e-07 0 9.18e-07 0.0001% Copper (+1 & +2) surface wate	Chlorine	surface water	6.36e-07	0	0	0	6.36e-07	3.79E-07
Chromiumsurface water08.84e-06008.84e-060.0005%Chromium (III)surface water6.23e-063.49e-0901.80e-096.23e-060.0004%Chromium (VI)surface water2.94e-072.33e-0701.80e-095.29e-073.15E-07Cobalt (Co I, Co II, Co III)surface water7.70e-100007.70e-104.59E-10CODsurface water7.67e-038.20e-020-2.69e-038.70e-025.1827%CODtreatment03.90e-02003.90e-022.3251%Colon bacillus (bacteria in large intestine)surface water03.89e-03003.89e-030.2320%Coppersurface water09.18e-07009.18e-070.0001%Copper (+1 & +2)surface water3.80e-062.84e-100-1.01e-113.80e-060.0002%	Chloroform	surface water	2.87e-12	0	0	1.74e-11	2.02e-11	1.21E-11
Chromium (III)surface water6.23e-063.49e-0901.80e-096.23e-060.0004%Chromium (VI)surface water2.94e-072.33e-0701.80e-095.29e-073.15E-07Cobalt (Co I, Co II, Co III)surface water7.70e-100007.70e-104.59E-10CODsurface water7.67e-038.20e-020-2.69e-038.70e-025.1827%CODtreatment03.90e-02003.90e-022.3251%Colon bacillus (bacteria in large intestine)surface water03.89e-03003.89e-030.2320%Coppersurface water09.18e-07009.18e-070.0001%Copper (+1 & +2)surface water3.80e-062.84e-100-1.01e-113.80e-060.0002%	Chromium	surface water	0	8.84e-06	0	0	8.84e-06	0.0005%
Chromium (VI)surface water2.94e-072.33e-0701.80e-095.29e-073.15E-07Cobalt (Co I, Co II, Co III)surface water7.70e-100007.70e-104.59E-10CODsurface water7.67e-038.20e-020-2.69e-038.70e-025.1827%CODtreatment03.90e-02003.90e-022.3251%Colon bacillus (bacteria in large intestine)surface water03.89e-03003.89e-030.2320%Coppersurface water09.18e-07009.18e-070.0001%Copper (+1 & +2)surface water3.80e-062.84e-100-1.01e-113.80e-060.0002%	Chromium (III)	surface water	6.23e-06	3.49e-09	0	1.80e-09	6.23e-06	0.0004%
Cobalt (Co I, Co II, Co III) surface water 7.70e-10 0 0 7.70e-10 4.59E-10 COD surface water 7.67e-03 8.20e-02 0 -2.69e-03 8.70e-02 5.1827% COD treatment 0 3.90e-02 0 0 3.90e-02 2.3251% Colon bacillus (bacteria in large intestine) surface water 0 3.89e-03 0 0 3.89e-03 0.2320% Copper surface water 0 9.18e-07 0 0.918e-07 0.0001% Copper (+1 & +2) surface water 3.80e-06 2.84e-10 0 -1.01e-11 3.80e-06 0.0002%	Chromium (VI)	surface water	2.94e-07	2.33e-07	0	1.80e-09	5.29e-07	3.15E-07
COD surface water 7.67e-03 8.20e-02 0 -2.69e-03 8.70e-02 5.1827% COD treatment 0 3.90e-02 0 0 3.90e-02 2.3251% Colon bacillus (bacteria in large intestine) surface water 0 3.89e-03 0 0 3.89e-03 0.2320% Copper surface water 0 9.18e-07 0 0 9.18e-07 0.0001% Copper (+1 & +2) surface water 3.80e-06 2.84e-10 0 -1.01e-11 3.80e-06 0.0002%	Cobalt (Co I, Co II, Co III)	surface water	7.70e-10	0	0	0	7.70e-10	4.59E-10
COD treatment 0 3.90e-02 0 0 3.90e-02 2.3251% Colon bacillus (bacteria in large intestine) surface water 0 3.89e-03 0 0 3.89e-03 0.2320% Copper surface water 0 9.18e-07 0 0 9.18e-07 0.0001% Copper (+1 & +2) surface water 3.80e-06 2.84e-10 0 -1.01e-11 3.80e-06 0.0002%	COD	surface water	7.67e-03	8.20e-02	0	-2.69e-03	8.70e-02	5.1827%
Colon bacillus (bacteria in large intestine) surface water 0 3.89e-03 0 0 3.89e-03 0.2320% Copper surface water 0 9.18e-07 0 0 9.18e-07 0.0001% Copper (+1 & +2) surface water 3.80e-06 2.84e-10 0 -1.01e-11 3.80e-06 0.0002%	СОР	treatment	0	3.90e-02	0	0	3.90e-02	2.3251%
intestine) surface water 0 3.89e-03 0 0 3.89e-03 0.2320% Copper surface water 0 9.18e-07 0 0 9.18e-07 0.0001% Copper (+1 & +2) surface water 3.80e-06 2.84e-10 0 -1.01e-11 3.80e-06 0.0002%	Colon bacillus (bacteria in large							
Copper surface water 0 9.18e-07 0 0 9.18e-07 0.0001% Copper (+1 & +2) surface water 3.80e-06 2.84e-10 0 -1.01e-11 3.80e-06 0.0002%	intestine)	surface water	0	3.89e-03	0	0	3.89e-03	0.2320%
Copper (+1 & +2) surface water 3.80e-06 2.84e-10 0 -1.01e-11 3.80e-06 0.0002%	Copper	surface water	0	9.18e-07	0	0	9.18e-07	0.0001%
	Copper (+1 & +2)	surface water	3.80e-06	2.84e-10	0	-1.01e-11	3.80e-06	0.0002%

Table J-14. LCD water outputs (wastewater and water pollutants) (kg/functional unit)

	A			I	/		% on Exaction of
Material	Disposition	Unstream	Mfø	Use	EOL	Total	76 OF Fraction of Total
Copper $(\pm 1 & \pm 2)$	treatment	0	4 28e-05	0	0	4 28e-05	0.0026%
Cvanide (-1)	surface water	4 49e-06	3.66e-06	0	-7.04e-13	8 15e-06	0.0005%
Cyanide (-1)	treatment	0	6.67e-07	0	0	6.67e-07	3 97E-07
Dichloromethane	surface water	8 22e-09	0.070 07	0	2 78e-11	8 25e-09	4 91E-09
Dissolved organic matter	Surface water	0.220 09	U	U	2.760 11	0.250 07	4.912 09
(chlorinated)	surface water	1.31e-06	0	0	0	1.31e-06	0.0001%
Dissolved organics	surface water	1.20e-03	4.67e-08	0	-8.34e-09	1.20e-03	0.0716%
Dissolved solids	surface water	3.21e-03	1.75e-01	0	-5.69e-05	1.78e-01	10.6251%
Edetic acid (EDTA)	surface water	9.46e-10	0	0	0	9.46e-10	5.63E-10
Ethylbenzene	surface water	1.43e-07	0	0	6.28e-11	1.43e-07	8.50E-08
Fluoride	surface water	3.39e-06	0	0	0	3.39e-06	0.0002%
Fluorides (F-)	surface water	5 14e-05	1 29e-02	0	-5.01e-06	1 30e-02	0 7724%
Fluorides (F-)	treatment	0	2.40e-04	0	0	2 40e-04	0.0143%
Formaldehyde	surface water	3 54e-14	2.400 04	0	0	3 54e-14	2 11E-14
Halogenated matter (organic)	surface water	1.65e-13	5.67e-12	0	-2 01e-13	5.64e-12	3 36E-12
Haveshloreethane	surface water	5.062.19	5.070-12	0	-2.010-15	5.060.18	2.01E-12
Havana	surface water	5.000-18	5 880 04	0	0	5.880.04	0.02519/
Hedroeine	sui lace water	4 25 - 10	5.886-04	0	0	1.25-10	0.033176
	surface water	4.556-10	0	0	0	4.556-10	2.39E-10
Hydrocarbons, remaining unspeciated	surface water	8.97e-05	1.70e-05	0	-7.87e-07	1.06e-04	0.0063%
Hydrochloric acid	treatment	0	3.29e-06	0	0	3.29e-06	0.0002%
Hypochlorite	surface water	8.62e-10	0	0	0	8.62e-10	5.14E-10
Hypochlorous acid	surface water	8.62e-10	0	0	0	8.62e-10	5.14E-10
Iodide (-1)	surface water	5.93e-07	0	0	0	5.93e-07	3.53E-07
Iron	surface water	0	1.31e-04	0	0	1.31e-04	0.0078%
Iron	treatment	0	8.33e-05	0	0	8.33e-05	0.0050%
Iron $(+2 \& +3)$	surface water	3.71e-04	1.73e-08	0	-1.02e-08	3.71e-04	0.0221%
Lead	surface water	0	8 18e-06	0	0	8 18e-06	0.0005%
Lead	treatment	0	8 33e-07	0	0	8 33e-07	4 96E-07
Lead cmpds	surface water	3.68e-06	5.67e-11	0	4 98e-10	3.68e-06	0.0002%
Lead cmpds	treatment	0	7.14e-06	0	4.960-10	7.14e-06	0.000276
Lithium salts	surface water	1869.11	7.140-00	0	0	1 869 11	2 80F 11
Magnesium (+2)	surface water	5 13e 04	0	0	0	5 13e 04	0.0305%
Manganasa	surface water	5.150-04	2 202 07	0	0	2 200 07	1.26E.07
Manganese minda	surface water	1 482 05	2.296-07	0	0	2.296-07	1.30E-07
Manuary	surface water	1.486-03	0 22 - 09	0	0	1.486-03	0.0009%
Mercury	treatment	0.0808	8.55e-08	0	0	8.556-08	4.90E-08
Mercury	surface water	9.086-08	9.098-08	0	1 (2 11	1.886-07	1.12E-07
Mercury compounds	surface water	5.82e-07	6.52e-14	0	1.62e-11	5.82e-07	3.4/E-0/
Metals, remaining unspeciated Molvbdenum (Mo II, Mo III, Mo IV,	surface water	3.55e-04	4.72e-04	0	-2.45e-05	8.03e-04	0.0478%
Mo V, Mo VI)	surface water	1.88e-06	0	0	0	1.88e-06	0.0001%
Morpholine	surface water	4.60e-09	0	0	0	4.60e-09	2.74E-09
Nickel	surface water	0	2.33e-07	0	0	2.33e-07	1.39E-07
Nickel	treatment	0	3.33e-06	0	0	3.33e-06	0.0002%
Nickel (+2)	surface water	4.39e-07	0	0	0	4.39e-07	2.62E-07
Nickel cmpds	surface water	3.29e-06	2.84e-11	0	-1.01e-12	3.29e-06	0.0002%
Nitrate	surface water	1.97e-05	3.15e-06	0	-1.27e-06	2.15e-05	0.0013%
Nitrates/nitrites	surface water	8.00e-05	0	0	0	8.00e-05	0.0048%
Nitrites	surface water	4,23e-07	0	0	0	4.23e-07	2.52E-07
Nitrogen	surface water	1.44e-05	7.98e-02	0	0	7.98e-02	4 7566%
	2.3. Tuee Water			0	0		1.750070

Table J-14. LCD water outputs (wastewater and water pollutants) (kg/functional unit)

Table J-14. LCD water outputs (wastewater and water pollutants) (kg/functional unit)

Matarial	Disposition	Unstroom	Mfa	Uas	FOI	Total	% or Fraction of
Material	Disposition	Opstream	1 26a 02	Use	EUL	1 26a 02	1 OLAI
Nitrogen		2.0404	1.20e-02	0	0	2.0404	0.7494%
	surface water	3.04e-04	5 262 04	0	0	5.04e-04	0.0181%
	surface water	2.186-00	3.500-04	0	0	2.6102	0.0321%
On a grease	treatment	0	3.01e-03	0	0	3.01e-03	0.2150%
Orthohoria agid	treatment	0	1.210.06	0	0	1.210.06	0.00019/
Other nitregen	treatment	2.25 - 04	7.662.10	0	2 72 11	2.25 - 04	0.0001%
Other argonias	surface water	2.236-04	7.000-10	0	-2.720-11	2.23e-04	0.0134%
Ounter organics	surface water	2.12e-05	0	0	0	2.12e-05	0.0013%
	surface water	1.896-09	0	0	1 27- 10	1.896-09	1.13E-09
o-xylene	surface water	1.50.05	1.76 04	0	1.376-10	1.37e-10	8.14E-11
Phenol Phasekata (PO42)	surface water	4.50e-05	1.76e-04	0	-6.22e-06	2.14e-04	0.0128%
Phosphate (PO43-)	surface water	1.11e-04	0	0	0	1.11e-04	0.0066%
Phosphates	surface water	2.23e-07	2.83e-08	0	7.27e-10	2.52e-07	1.50E-07
Phosphorus (yellow or white)	surface water	1.92e-06	4.33e-03	0	0	4.33e-03	0.2581%
Phosphorus (yellow or white)	treatment	0	6.91e-03	0	0	6.91e-03	0.4114%
Phosphorus pentoxide	surface water	7.63e-09	0	0	0	7.63e-09	4.54E-09
Polychlorinated biphenyls	surface water	0	1.14e-08	0	0	1.14e-08	6.80E-09
Polycyclic aromatic hydrocarbons	surface water	2.87e-06	6.81e-11	0	-2.41e-12	2.87e-06	0.0002%
Potassium (+1)	surface water	2.46e-04	0	0	0	2.46e-04	0.0146%
Rubidium ion (Rb+)	surface water	5.93e-08	0	0	0	5.93e-08	3.53E-08
Salts (unspecified)	surface water	2.54e-04	7.84e-05	0	-2.78e-06	3.29e-04	0.0196%
Sand	surface water	2.95e-09	0	0	0	2.95e-09	1.76E-09
Saponifiable oils and fats	surface water	7.30e-05	0	0	0	7.30e-05	0.0044%
Selenium	surface water	2.99e-06	0	0	2.88e-11	2.99e-06	0.0002%
Silver compounds	surface water	3.56e-09	0	0	5.26e-10	4.09e-09	2.43E-09
Sodium (+1)	surface water	1.73e-01	3.41e-01	0	-2.03e-02	4.94e-01	29.4192%
Strontium (Sr II)	surface water	9.60e-05	0	0	0	9.60e-05	0.0057%
Sulfate ion (-4)	surface water	2.40e-02	2.94e-03	0	-1.20e-06	2.69e-02	1.6033%
Sulfate ion (-4)	treatment	0	1.32e-04	2.55e-02	4.84e-06	2.57e-02	1.5285%
Sulfide	surface water	2.20e-06	9.86e-09	0	-1.74e-09	2.20e-06	0.0001%
Sulfites	surface water	4.80e-07	0	0	0	4.80e-07	2.86E-07
Sulfur	surface water	1.04e-11	0	0	0	1.04e-11	6.18E-12
Suspended solids	surface water	4.98e-03	5.80e-02	0	-1.44e-03	6.15e-02	3.6643%
Suspended solids	treatment	0	5.60e-03	6.65e-04	1.26e-07	6.26e-03	0.3732%
Tars (unspecified)	surface water	5.06e-15	0	0	0	5.06e-15	3.01E-15
Tetrachloroethylene	surface water	1.26e-14	2.29e-08	0	1.74e-11	2.29e-08	1.37E-08
Tin	surface water	0	4.58e-07	0	0	4.58e-07	2.73E-07
Tin (Sn++, Sn4+)	surface water	5.11e-05	0	0	0	5.11e-05	0.0030%
Titanium tetrachloride	surface water	3.52e-05	0	0	0	3.52e-05	0.0021%
TOCs	surface water	8.96e-04	4.25e-08	0	-1.51e-09	8.96e-04	0.0534%
Toluene	surface water	1.60e-06	6.24e-10	0	5.83e-10	1.60e-06	0.0001%
Trichloroethylene	surface water	7.66e-13	2.29e-08	0	1.74e-11	2.29e-08	1.37E-08
Triethylene glycol	surface water	1.86e-06	0	0	0	1.86e-06	0.0001%
Vanadium (V3+, V5+)	surface water	3.18e-06	0	0	0	3.18e-06	0.0002%
Vinyl chloride	surface water	0	0	0	3.48e-11	3.48e-11	2.07E-11
VOCs, remaining unspeciated	surface water	2.07e-06	0	0	0	2.07e-06	0.0001%
Waste metals, unspecified	surface water	9.37e-05	0	0	0	9.37e-05	0.0056%
Waste oil	surface water	1.75e-03	4.87e-03	0	-2.06e-04	6.41e-03	0.3817%
Xylene (mixed isomers)	surface water	6.41e-06	0	0	1.76e-10	6.41e-06	0.0004%
Zinc (+2)	surface water	7.41e-06	1.40e-09	0	-2.09e-10	7.41e-06	0.0004%
Zinc (elemental)	surface water	0	2.63e-06	0	0	2.63e-06	0.0002%
Total water pollutant	s	4.60e-01	1.23e+00	2.62e-02	-4.09e-02	1.68e+00	100.00%

	Total
Acetic acid landfill 0 4.46E-03 0 0 4.46E-03	7.09E-04
Acetone treatment 0 2.77E-02 0 0 2.77E-02	4.40E-03
Acid waste (D002 waste) treatment 0 1.19E-03 0 0 1.19E-03	1.89E-04
Barium debris (D008 waste) landfill 0 9.91E-06 0 0 9.91E-06	1.57E-06
Chrome debris (D007 waste) treatment 0 6.83E-06 0 0 6.83E-06	1.08E-06
Chrome liquid waste (D007 waste) recycling/reuse 0 4.54E-04 0 0 4.54E-04	7.22E-05
Chromium landfill 0 1.52E-06 0 0 1.52E-06	2.41E-07
EOL LCD Monitor, landfilled landfill 0 0 0 1.64E+00 1.64E+00	26.13%
Ferric chloride recycling/reuse 0 1.37E-02 0 0 1.37E-02	2.18E-03
Flammable liquids (F003 waste)treatment09.13E-04009.13E-04	1.45E-04
Hazardous waste, unspecified landfill 6.72E-03 2.97E-02 0 -1.05E-03 3.54E-02	5.62E-03
Hazardous waste, unspecified recycling/reuse 0 1.42E-02 0 0 1.42E-02	2.26E-03
Hazardous waste, unspecified treatment 0 6.16E-02 0 0 6.16E-02	9.78E-03
HCFC-225ca recycling/reuse 0 3.11E-05 0 0 3.11E-05	4.94E-06
HCFC-225cb recycling/reuse 0 3.11E-05 0 0 3.11E-05	4.94E-06
Hydrofluoric acid landfill 0 8.24E-05 0 0 8.24E-05	1.31E-05
Isopropyl alcohol recycling/reuse 0 1.69E-01 0 0 1.69E-01	2.69%
Isopropyl alcohol treatment 0 1.91E+00 0 0 1.91E+00	30.41%
Mercury recycling/reuse 0 2.00E-06 0 0 2.00E-06	3.18E-07
Nitric acid landfill 0 3.43E-04 0 0 3.43E-04	5.45E-05
Phosphoric acid landfill 0 1.44E-02 0 0 1.44E-02	2.29E-03
PWB-Decontaminating debris treatment 0 6.85E-03 0 0 6.85E-03	1.09E-03
PWB-Lead contaminated waste oil treatment 0 5.14E-03 0 0 5.14E-03	8.16E-04
PWB-Route dust recycling/reuse 0 5.31E-03 0 0 5.31E-03	8.43E-04
PWB-Solder dross recycling/reuse 0 2.96E-02 0 0 2.96E-02	4.70E-03
PWB-Waste cupric etchant recycling/reuse 0 9.93E-02 0 0 9.93E-02	1.58%
Remover, unspecified recycling/reuse 0 8.84E-02 0 0 8.84E-02	1.40%
Remover, unspecified treatment 0 3.03E-01 0 0 3.03E-01	4.81%
Rinse, unspecified recycling/reuse 0 4.67E-02 0 0 4.67E-02	7.42E-03
Silver landfill 0 2.72E-09 0 0 2.72E-09	4.33E-10
Sodium sulfate recycling/reuse 0 2.44E-01 0 0 2.44E-01	3.89%
Spent solvent (non-halogenated) treatment 0 4.66E-02 0 0 4.66E-02	7.41E-03
Spent solvent (with halogenated materials) treatment 0 1.55E-02 0 0 1.55E-02	2.47E-03
Spent solvents (F003 waste) treatment $0 2.74E-04 0 0 2.74E-04$	4.35E-05
Tetramethyl ammonium hydroxide recycling/reuse $0 142E-01 0 0 142E-01$	2.26%
Thinner, unspecified treatment 0 5.40E-01 0 0 5.40E-01	8.57%
Unspecified sludge land (other than 0 3.09E-02 0 0 3.09E-02	4.91E-03
Waste acid (chrome mixed acid) recycling/reuse 0 718E-03 0 0 718E-07	1 14E-03
Waste acid (mainly HF) recycling/reuse $0.569E-01$ $0.0569E-01$	9.04%
Waste acid (mainly HF) treatment 0 1.36E-01 0 0 1.36E-01	2.15%
Waste acid (mostly 3% HCl solution) recycling/reuse 0 1 82E-04 0 0 1 82E-04	2 89E-05
Waste acids unspecified recycling/reuse $0.324F_{-}02$ $0.0324F_{-}02$	5 15E-03
Waste Batch (Ba Pb) (D008 waste) landfill $0 = 6.55E-05$ $0 = 0.655E-05$	1.04E-05
Waste Datein (Da, 10) (Dood waste) in and in 0 0 0 0 0 0 0 0 0 0	1.04E-05
Waste CCFL, with near treatment $0 \times 17E \cdot 10 = 0 \times 17E \cdot 10$	1.30E-08
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	1.50E-10
waste grass, with increase 0 $1.03E-10$ 0 0 $1.03E-10$ Waste metals unspecifiedrecycling/reuse 0 $1.17E_02$ 0 0 $1.17E_02$	1.0/E-11
waste colvent (photoresist) recycling/reuse $0.205E02$ $0.00205E02$	1.0JE-04
w asic solvent (photoresist)recycling/reuse 0 $2.03E-02$ 0 0 $2.17E-02$ Waste solvent (photoresist)treatment 0 $2.17E-02$ 0 0 $2.17E-02$	3.20E-03
Total hazardous waste 6 72E-03 4 64E+00 0 00E+00 1 64E+00 6 29E+00	100.00%

Table J-15.	LCD hazardous	waste outputs	(kg/functional unit)
-------------	---------------	---------------	----------------------

	e J-10. LCD	sond waste	e outputs ((kg/lunc	lional unit)	
Material	Disposition	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
abrasive sludge	recycling/reuse	0	1.95E-03	0	0	1.95E-03	3.73E-05
acid absorbent	landfill	0	3.77E-06	0	0	3.77E-06	7.20E-08
Aluminum (elemental)	landfill	1.23E-05	0	0	0	1.23E-05	2.35E-07
Aluminum scrap	recycling/reuse	2.55E-07	8.77E-06	0	9.52E-03	9.53E-03	1.82E-04
Aluminum scrap, Wabash 319	recycling/reuse	0	2.37E-08	0	0	2.37E-08	4.53E-10
Arsenic	landfill	4.91E-09	0	0	0	4.91E-09	9.38E-11
Bauxite residues	landfill	1.70E-05	5.87E-04	0	-1.96E-05	5.84E-04	1.12E-05
blasting media	landfill	0	1.70E-05	0	0	1.70E-05	3.24E-07
Broken CCFL	landfill	0	2.69E-07	0	0	2.69E-07	5.14E-09
Cadmium	landfill	4.92E-10	0	0	0	4.92E-10	9.40E-12
Calcium	landfill	4.91E-05	0	0	0	4.91E-05	9.38E-07
Carbon	landfill	3.79E-05	0	0	0	3.79E-05	7.24E-07
CARBON STEEL SCRAP	recvcling/reuse	0	0	0	4.58E-01	4.58E-01	8.75E-03
Cardboard	treatment	0	1.82E-05	0	0	1.82E-05	3.47E-07
Chromium (III)	landfill	3.07E-08	0	0	0	3.07E-08	5.87E-10
Chromium (VI)	landfill	3.07E-08	0	0	0	3.07E-08	5.87E-10
Cinders from LCD glass mfg	landfill	0	3 83E-04	0	0	3 83E-04	7 31E-06
Coal waste	landfill	0	2 28E+00	1 90E+01	3 60E-03	2 13E+01	40.64%
Cobalt	landfill	2 36E-10	0	0	0	2.15E+01	4 51E-12
Cobalt nitrate	treatment	2.501 10	2 83E-06	0	0	2.50E 10	5.40F-08
Copper	landfill	1 18F-09	2.051_00	0	0	1.18E-09	2.26E-11
Diesel fuel	treatment	1.18E-09	1 88E 06	0	0	1.18E-09	3.60E.08
Dust	treatment	0	1.50E-00	0	0	1.50E-00	3.03E.06
Dust	londfill	0	9.90E.01	7 24E±00	1 20E 02	8 22E±00	15 72%
EQL LCD Monitor incinerated	treatment	0	0.001-01	7.54E+00	0.75E.01	0.25E+00	1 9 6 9 /
EOL LCD Monitor, included	londfill	0	0	0	9.73E-01	9.73E-01	1.3070
EOL LCD Monitor, rangeled	ranuling/rausa	0	0	0	0.75E 01	0.75E 01	1.7170
EOL LCD Monitor, recycled	recycling/reuse	0	0	0	9.73E-01	9.73E-01	1.80%
EOL LCD Monitor, remanufactured	recycling/reuse	2.45E.02	1.005.02	0	9.73E-01	9.73E-01	1.80%
FGD studge	landiili	2.45E-02	1.09E-02	4.75E+00	-0.11E-04	5.48E-02	0.04E-04
Fly/bottom asn	landiili	0 2.4(E.05	5.70E-01	4./3E+00	9.01E-04	5.52E+00	10.10%
Iron	landiii	2.40E-05	0	0	1.105+00	2.40E-05	4.70E-07
Iron scrap	recycling/reuse	1.6/E-01	0	0	1.10E+00	1.27E+00	2.42%
Isopropyl alconol	recycling/reuse	0	2.53E-02	0	0	2.53E-02	4.84E-04
Isopropyl alconol	treatment	0	1.03E-02	0	0	1.03E-02	1.9/E-04
LCD glass	recycling/reuse	0	0	0	8.77E-02	8.77E-02	1.68E-03
LCD glass EP dust	landfill	0	4.77E-05	0	0	4.77E-05	9.12E-07
LCD glass EP dust	recycling/reuse	0	2.32E-04	0	0	2.32E-04	4.43E-06
LCD glass, unspecified	landfill	0	1.13E-03	0	0	1.13E-03	2.15E-05
LCD panel waste	landfill	0	2.43E-02	0	0	2.43E-02	4.64E-04
Lead	landfill	5.42E-09	0	0	0	5.42E-09	1.04E-10
Manganese	landfill	4.91E-07	0	0	0	4.91E-07	9.38E-09
Mercury	landfill	3.33E-11	0	0	0	3.33E-11	6.36E-13
Mineral waste	landfill	2.20E-01	1.26E-04	0	-4.46E-06	2.21E-01	4.21E-03
Mining waste	landfill	1.41E-01	0	0	-1.23E-06	1.41E-01	2.69E-03
Mixed industrial (waste)	landfill	4.34E-02	4.83E-02	0	-1.35E-03	9.04E-02	1.73E-03
Nickel	landfill	1.77E-09	0	0	0	1.77E-09	3.38E-11
Nickel nitrate	treatment	0	2.83E-06	0	0	2.83E-06	5.40E-08
Nitrogen	landfill	9.83E-09	0	0	0	9.83E-09	1.88E-10
Non mineral waste (inert)	landfill	9.80E-06	0	0	0	9.80E-06	1.87E-07
Non toxic chemical waste (unspecified)	landfill	5.72E-03	2.95E-05	0	-1.05E-06	5.75E-03	1.10E-04
Oily rags & filter media	landfill	0	1.51E-05	0	0	1.51E-05	2.88E-07
Oily rags & filter media	recycling/reuse	0	1 88E-06	0	0	1 88E-06	3 60E-08

Table I 16 I CD solid waste outputs (l/a/functional unit)

parts cleaner solvent recycling/reuse 0 3.77E-06 0 0 0 3.77E-06 7.20E-08 Phosphonstx (yellow or white) Inaffill 6.29E-07 0 0 6.29E-07 1.20E-08 Polycator residudge Inaffill 0 1.52E-05 0 0 3.50E-01 7.46E-03 Polycator resin recycling/reuse 0 3.20E-02 0 0 2.52E-05 2.06E-03 Polycathylene/olypropylene waste treatment 0 2.72E-03 0 0 7.53E-03 2.01E-06 1.43E-04 Prinatel wiring board (PWB) Inaffill 0 7.59E-03 0 0 3.55E-05 0 0 0 3.55E-05 2.01E-00 3.55E-00 3.04E-00 2.95E-00 3.04E-00 2.95E-00 3.35B, 3.35B, 3.35E-01 3.02E-00 3.04E-00 2.95E-00 3.04E-00 2.95E-00 3.04E-00 2.95E-00 3.04E-00 2.95E-00 3.04E-00 2.95E-00 3.04E-00 2.95E-00 3.04E-00 3.05E-00 3.05E-00 3	Material	Disposition	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Phosphones (yellow or white) Indfill 6.29E-07 0 0 6.29E-07 1.20E-08 Phating process sludge Indfill 0 1.52E-05 0 0.30E-01 1.52E-05 2.90F-07 Polyestrersin reccycling/reuse 0 3.20E-02 0 0 2.27E-03 0 0 2.72E-03 0 0 7.55E-03 2.90E-04 1.72E-04 1.72E-04 1.72E-04 1.72E-04 1.72E-04 0 0 1.75E-04 1.72E-04 1.72E-0	parts cleaner solvent	recycling/reuse	0	3.77E-06	0	0	3.77E-06	7.20E-08
Plating process sludgeIndIII01.52E-05003.90E-013.90E-017.46E-03Polycarhonatereccycling/reuse03.20E-0203.20E-021.6E-047.46E-03Polychlycher, foamedreatsment03.29E-04003.20E-021.91E-05Polychlycher, foamedreatsment02.27E-03002.52E-035.20E-05Polychlycher, foamedreatsment03.29E-04001.53E-042.91E-06Phinted wiring board (PMB)landfill06.59E-030005.90E-04PMB-Drill dixtlandfill3.29E-053.09E-020003.52E-055.90E-04Remover, unspecified)reatsment3.09E-023.09E-020003.52E-055.90E-04Slag and ashlandfill8.19E-023.49E-020008.13E-042.19E-03Slag and ashlandfill04.06E-050008.13E-042.19E-06Studge (calcin fluctic), Cal ² 2recycling/reuse3.23E+040003.53E+046.22E+06Studge (calcin fluctic), Cal ² 2recycling/reuse03.32E+040003.54E+042.40E+06Studge (calcin fluctic), Cal ² 2recycling/reuse03.32E+040003.54E+042.40E+06Studge (calcin fluctic), Cal ² 2recycling/reuse03.32E+04000	Phosphorus (yellow or white)	landfill	6.29E-07	0	0	0	6.29E-07	1.20E-08
Polycapkanate recycling/reuse 0 3.90E-01 3.90E-01 7.46E-03 Polyctstre resin recycling/reuse 0 3.20E-02 0 0.099E-04 0.0 3.20E-02 6.11E-04 Polyctstre, foamed treatment 0 2.72E-03 0 0.0 2.72E-03 5.20E-05 Polysthylene/polyropylene waste treatment 0 2.72E-03 0 0.0 7.50E-03 1.43E-04 Printed wiring board (PWB) landfill 0 7.50E-03 0 0.0 0.59E-03 1.43E-04 PWB-Drill dust tandfill 3.25E-05 0 0 0 3.26E-05 5.90E-04 Skag and ash recycling/reuse 8.02E+00 3.40E+00 0 9.25E+04 3.25E-04 0 0 3.25E+04 2.91E-06 Stalag and ash landfill 0 4.36E-03 0 0 3.25E+04 0.0 0 7.37E+02 0 0 3.25E+04 2.91E-06 Stalage from 1.0216 rccycling/reuse <	Plating process sludge	landfill	0	1.52E-05	0	0	1.52E-05	2.90E-07
Polyester resinrecycling/reuse03.20E-02003.20E-026.11E-04Polyets/here, fooptreatment09.99E-04009.99E-041.91E-05Polyets/here, foopreatment01.35E-04007.27E-035.20E-05Polyets/here, fooplandfill01.35E-04001.53E-042.91E-06Phinted wiring board (PWB)landfill06.59E-03006.59E-031.43E-04Remover, unspecifiedtreatment03.09E-020003.25E-055.90E-04Stag and ashrecycling/reuse8.02E+003.40E+0009.67E+001.75E+003.35%Stag and ashrecycling/reuse8.02E+008.34E+020008.13E-041.55E-05Stadge (calcium fluoride, Cal ²)recycling/reuse3.23E-040001.53E-042.91E-06Stadge form LCD glass mfglandfill01.53E-040003.23E-046.22E+06Stel carpot (inplated)recycling/reuse7.73E-020007.73E-020007.73E-02Stel for any (inplated)recycling/reuse7.73E-0200007.73E-0200007.73E-0200007.73E-0200007.73E-0200007.73E-0200007.73E-020 <td>Polycarbonate</td> <td>recycling/reuse</td> <td>0</td> <td>0</td> <td>0</td> <td>3.90E-01</td> <td>3.90E-01</td> <td>7.46E-03</td>	Polycarbonate	recycling/reuse	0	0	0	3.90E-01	3.90E-01	7.46E-03
Polychychene, foamedreatment09.99E-04009.99E-041.91E-05Polychychene/polynene/polynene wastetreatment02.72E-03002.72E-035.20E-05Poltassiun (Zhornate)landfill07.50E-030007.50E-031.43E-04Phinted wiring board (PWB)landfill06.59E-031.00E03.09E-025.90E-04Remover, unspecifiedtreatment00.90F-02000.92E-055.20E-05Slag and ashrecycling/reuse8.02E-003.09E-0209.67E+001.75E+033.35%Slag and ashrecycling/reuse08.13E-04009.67E+001.75E+042.19E-03Sudge (calcium fluoride, CaP2)recycling/reuse08.13E-040001.55E-055.000000.77E+072.19E-03Soldge from LCD glass miglandfill01.53E-040001.53E-042.91E-062.91E-022.91E-062	Polyester resin	recycling/reuse	0	3.20E-02	0	0	3.20E-02	6.11E-04
Polyethylencipolypropylene wastetreatment02.72E-03002.72E-035.20E-05Potassim CarbonateInadfill01.53E-04001.53E-042.91E-06Printed wing board (PWB)Iandfill06.59E-030003.29E-025.90E-04Remover, unspecifiedIreatiment03.09E-020003.25E-0500003.25E-0500003.25E-056.22E-077.50E-030.325E-056.22E-077.50E-033.35%5.325K3.35%5.325K3.35%5.325K3.35%5.325K000003.25E-041.55E-055.046E7.77E-075.046E00001.55E-042.91E-062	Polyethylene, foamed	treatment	0	9.99E-04	0	0	9.99E-04	1.91E-05
Potassium Carbonate landfill 0 1.53E-04 0 0 1.53E-04 2.91E-06 Printed wiring board (PWB) landfill 0 7.50F-03 0 0 6.59E-03 0 6.59E-03 0 6.59E-03 0 6.59E-03 0 0 3.09E-02 5.90E-04 Remover, unspecified treatment 0 3.40E+00 0 9.67E+00 3.33E+05 6.22E+07 Slag and ash landfill 8.19E-02 3.40E+02 0 -1.99E+03 1.15E+01 2.19E+03 Sludge (calcium fluoride, CaF2) recycling/reuse 0 8.13E+04 0 0 8.13E+04 1.55E+05 Soldge from LCD glass mig landfill 0 1.53E+04 0 0 0 7.77E+07 2.01E+06 5.25E+04 6.22E+06 5.25E+04 6.22E+06 5.25E+04 6.22E+06 5.25E+04 0 0 7.73E+02 1.48E+04 1.45E+04 2.40E+04 2.40E+04 2.40E+04 2.40E+04 2.40E+04 1.45E+04 2.40E+04	Polyethylene/polypropylene waste	treatment	0	2.72E-03	0	0	2.72E-03	5.20E-05
Printed wiring board (PWB)landfill07.50E-03007.50E-031.43E-04PWB-Drill dustInandfill06.59E-03003.09E-025.90E-04Sevage shudge (unspecified)landfill3.25E-050003.25E-056.22E-07Slag and ashrecycling/reuse8.02E+003.40E+000-9.67E+001.15E-102.19E-03Slag and ashlandfill8.02E+008.40E+050-1.99E-031.15E-102.19E-03Sladge forn LCD glass mfglandfill04.06E+05004.06E+057.77E-07Sodium Carbonatelandfill01.53E-440001.53E-042.91E-06Stanton sludgerecycling/reuse7.73E+020007.73E-061.48E-07Unspecified nonhaz/ndous wasterecycling/reuse01.26E-042.35E-040007.73E-061.48E-07Unspecified sludgerecycling/reuse01.26E-042.35E-040003.56E-04003.56E-04003.56E-04003.56E-04003.56E-040003.56E-04003.56E-04003.56E-04003.56E-04003.56E-04003.56E-04003.56E-04003.56E-04003.56E-04003.56E-04003.56E-0400 <td>Potassium Carbonate</td> <td>landfill</td> <td>0</td> <td>1.53E-04</td> <td>0</td> <td>0</td> <td>1.53E-04</td> <td>2.91E-06</td>	Potassium Carbonate	landfill	0	1.53E-04	0	0	1.53E-04	2.91E-06
PWB-Drill dist landfill 0 6.59E-03 0 0 6.59E-03 1.26E-04 Remover, unspecified treatment 0 3.09E-02 0 0 3.09E-02 5.09E-04 Slag and ash recycling/reuse 8.02E+00 3.40E+00 0 -9.67E+00 1.75E+00 3.35% Slag and ash landfill 8.19E-04 0 0 8.13E+04 0 0 3.35*04 2.19E+03 Statiout 1.53E+04 0 0 0 8.13E+04 0 0 0 8.13E+04 0 0 0 3.55*04 6.22E+06 Statiout Statiout 1.26E+04 0 0 0 1.26E+04 2.40E+06 0 0 1.26E+04 2.40E+05 0 0 1.26E+04 2.40E+05	Printed wiring board (PWB)	landfill	0	7.50E-03	0	0	7.50E-03	1.43E-04
Remover, unspecifiedtreatment03.09E-02003.09E-025.90E-04Sewage sludge (unspecified)landfill3.25E-058.02E+003.25E+056.22E+07 3.35% Slag and ashrecycling/reuse8.02E+003.40E+009.76E+001.75E+01 3.35% Slag and ashlandfill8.19E+02 $3.49E+02$ 0 $-1.99E+03$ $1.15E+01$ $2.19E+03$ Sladge fcank flow fLCD glass mfglandfill0 $8.13E+04$ $0.55E+04$ 00 $6.32E+07$ Sodium Carbonatelandfill0 $1.55E+04$ 00 $0.53E+04$ $6.22E+06$ Stantons sludgerecycling/reuse $3.25E+04$ 00 $0.325E+04$ $6.22E+06$ Stantons sludgerecycling/reuse $7.73E+02$ 00 $7.73E+02$ $1.48E+03$ Unspecified nonhazardous wasterecycling/reuse0 $1.26E+04$ 00 $3.56E+04$ $2.40E+06$ Unspecified sludgerecycling/reuse0 $8.46E+01$ $0.573E+02$ $0.573E+02$ $1.09E+03$ $3.67E+04$ Unspecified sludgetreatment0 $0.573E+02$ $0.68E+05$ $3.73E+02$ $1.09E+03$ $3.27E+04$ Unspecified sludgetreatment0 $0.62E+04$ 0 $0.71E+03$ $3.27E+04$ Unspecified sludgetreatment0 $0.62E+04$ 0 $0.71E+03$ $3.27E+04$ Unspecified sludgetreatment0 $0.62E+04$ 0 $0.71E+03$ $3.27E+04$ Unspecified sludw	PWB-Drill dust	landfill	0	6.59E-03	0	0	6.59E-03	1.26E-04
Sewage sludge (unspecified) landfill $3.25E-05$ 0 0 $0.25E-05$ $6.22E-07$ Slag and ash recycling/reuse $8.02E+00$ $3.40E+00$ 0 $9.57E+00$ $1.75E+00$ 3.35^{96} Slag and ash landfill $8.19E-04$ 0 0 $8.13E-04$ 0	Remover, unspecified	treatment	0	3.09E-02	0	0	3.09E-02	5.90E-04
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Sewage sludge (unspecified)	landfill	3.25E-05	0	0	0	3.25E-05	6.22E-07
Sing and ashIandfill $8.19E-02$ $3.49E-02$ 0 $-1.99E-03$ $1.15E-01$ $2.19E-03$ sludge (calcium fluoride, CaF2)recycling/reuse0 $8.13E-04$ 00 $8.13E-04$ 1.55E-04 $2.91E-06$ Sludge from LCD glass mfglandfill0 $1.53E-04$ $2.91E-06$ $3.25E-04$ 00 $1.53E-04$ $2.91E-06$ Stannous sludgerecycling/reuse $3.25E-04$ 000 $3.25E-04$ $6.22E-06$ Steel scrap (inplated)recycling/reuse $7.73E-02$ 00 $7.73E-02$ $1.48E-03$ Unspecified nonhazardous wasterecycling/reuse0 $1.26E-04$ 00 $3.56E-04$ 0 0 $3.56E-04$ $6.80E-06$ Unspecified sludgerecycling/reuse0 $8.46E-01$ 0 0 $8.46E-01$ 1.62% Unspecified sludgetreatment0 $5.73E-02$ 00 $7.71E-03$ $2.04E-06$ Unspecified sludgetreatment0 $1.63E+00$ 00 1.62% 1.62% Unspecified solid wastetreatment0 $1.63E+00$ 0 0 1.62% Unspecified solid wastetreatment0 $1.63E+00$ 0 0 $1.71E+03$ $2.04E-05$ Unspecified solid wastetreatment 0 $1.63E+00$ 0 0 $1.71E+03$ $2.04E-05$ Unspecified solid wastetreatment 0 $1.62E+04$ 0 $-3.84E-05$ $1.07E-03$ $2.04E-05$ Unspecified solid	Slag and ash	recycling/reuse	8.02E+00	3.40E+00	0	-9.67E+00	1.75E+00	3.35%
studge (calcium fluoride, CaF2)recycling/reuse0 $8.13E-04$ 00 $8.13E-04$ $1.55E-05$ Shudge from LCD glass mfglandfill0 $4.06E-05$ 00 $4.06E-05$ $7.77E-07$ Sodium Carbonatelandfill0 $1.53E-04$ 00 $1.53E-04$ $2.91E-06$ Stannous shudgerecycling/reuse $3.25E-04$ 000 $3.25E-04$ $6.22E-06$ Stell scrap (tinplated)recycling/reuse $7.75E-02$ 000 $7.73E-02$ $1.48E-03$ Sulfarlandfill $7.37E-06$ 000 $7.37E-06$ $1.48E-03$ Unspecified onhazardous wasterecycling/reuse0 $8.46E-01$ 00 $8.46E-01$ $1.26E-04$ 00 $1.26E-04$ 0.0 $1.26E-04$ 0.0 $1.26E-04$ 0.0 $1.26E-04$ 0.0 $0.5.73E-02$ $1.09E-03$ 1	Slag and ash	landfill	8.19E-02	3.49E-02	0	-1.99E-03	1.15E-01	2.19E-03
Shudge from LCD glass mfg Iandfill 0 4.06E-05 0 0 4.06E-05 7.77E-07 Sodium Carbonate Iandfill 0 1.53E-04 0 0 3.25E-04 2.91E-06 Stannous sludge recycling/reuse 3.25E-04 0 0 0 7.73E-07 Steel scrap (tiplated) recycling/reuse 7.73E-06 0 0 7.73E-06 1.41E-07 Unspecified nonhazardous waste recycling/reuse 0 1.26E-04 0 0 3.56E-04 6.80E-01 0 0 3.56E-04 6.80E-06 0 0 5.73E-02 1.09E-03 1.95E-01 2.95E-01 1.95E-0	sludge (calcium fluoride, CaF2)	recvcling/reuse	0	8.13E-04	0	0	8.13E-04	1.55E-05
Odium Carbonate Iandfill 0 1.53E-04 0 0 1.53E-04 2.91E-06 Stamous sludge recycling/reuse 3.25E-04 0 0 0 3.25E-04 6.22E-06 Steel scap (tinplated) recycling/reuse 7.73E-02 0 0 0 7.73E-02 1.48E-03 Suthur Iandfill 7.37E-06 0 0 0 7.37E-06 1.41E-07 Unspecified nonhazardous waste recycling/reuse 0 8.46E-01 0 0 8.46E-01 6.80E-04 0 0 5.73E-02 0 0 5.73E-02 0 0 5.73E-02 1.09E-01 1.89E+00 3.22% Unspecified solid waste Iandfill 2.40E+00 0 0 5.73E-02 1.0 0 1.71E+00 3.27% Unspecified solid waste treatment 0 1.63E+00 0 0 1.71E+00 3.27% Unspecified solid waste treatment 4.01E-04 6.44E-04 0 3.84E-05 1.07E-03 2.	Sludge from LCD glass mfg	landfill	0	4.06E-05	0	0	4.06E-05	7.77E-07
Stannous sludgerecycling/reuse $3.25E-04$ 0 0 0 $3.25E-04$ $6.22E-06$ Steel scrap (tinplated)recycling/reuse $7.73E-02$ 0 0 0 $7.73E-02$ $1.48E-03$ Sulfurlandfill $7.37E-06$ 0 0 0 $7.73E-02$ $1.48E-03$ Sulfurlandfill 0 $3.56E-04$ 0 0 0 $1.26E-04$ $2.40E-06$ Unspecified onhazardous wasterecycling/reuse 0 $8.46E-01$ 0 0 $8.46E-01$ $1.62E-04$ Unspecified sludgetreatment 0 $5.73E-02$ 0 0 $5.73E-02$ $1.09E-03$ Unspecified solid wastelandfill $2.40E+00$ 0 0 $5.73E-02$ $1.09E-03$ Unspecified solid wastetreatment 0 $1.63E+00$ 0 0 $1.63E+00$ 3.11% Unspecified solid wastetreatment 0 $1.63E+00$ 0 0 $1.63E+00$ 3.11% Unspecified solid wastetreatment $4.0E-04$ 0 0 $9.74E-02$ $-1.86E-03$ Unspecified wastelandfill 0 0 0 0 $2.70E-01$ $1.08E-02$ Unspecified wastelandfill 0 0 0 0 0 $1.62E-04$ Unspecified wastelandfill 0 0 0 0 0 $1.62E-04$ Waste akiali, color filter developer,recycling/reuse 0 $8.91E-02$ 0 0 0 $1.52E-03$ <	Sodium Carbonate	landfill	0	1.53E-04	0	0	1.53E-04	2.91E-06
Steel scrap (tinplated) recycling/reuse 7.73E-02 0 0 7.73E-02 1.48E-03 Sulfur landfill 7.37E-06 0 0 0 7.37E-06 1.41E-07 Unspecified nonhazardous waste recycling/reuse 0 1.26E-04 0 0 3.56E-04 2.40E-06 Unspecified sludge recycling/reuse 0 8.46E-01 0 0 8.46E-01 1.62% Unspecified sludge treatment 0 5.73E-02 0 0 5.73E-02 1.09E-03 Unspecified solid waste treatment 0 5.73E-02 0 0 1.71E+00 3.22% Unspecified solid waste treatment 0 1.63E+00 0 0 1.63E+00 3.11% Unspecified solid waste treatment 0 1.63E+00 0 0 1.63E+00 3.11% Unspecified solid waste treatment 0 1.62E+04 0 -3.84E+05 1.07E-03 2.04E+05 Unspecified waste landfill	Stannous sludge	recycling/reuse	3.25E-04	0	0	0	3.25E-04	6.22E-06
Sulfur landfill 7.37E-06 0 0 7.37E-06 1.41E-07 Unspecified nonhazardous waste recycling/reuse 0 1.26E-04 0 0 3.56E-04 6.80E-06 Unspecified sludge landfill 0 3.56E-04 0 0 3.56E-04 6.80E-06 Unspecified sludge recycling/reuse 0 8.46E-01 0 0 8.46E-01 1.62% Unspecified solid waste landfill 2.40E+00 0 0 5.73E-02 1.09E-03 Unspecified solid waste recycling/reuse 1.50E+00 2.11E-01 0 0 1.63E+00 3.17% Unspecified solid waste treatment 4.61E-04 6.44E-04 0 -3.84E-05 1.07E-03 2.04E-05 Unspecified vaste landfill 0 0 0 9.74E-02 -9.74E-02 -1.86E-03 Unspecified vaste landfill 0 2.70E-01 0 6.22E-04 1.07E-01 0 0.27E-01 5.16E-03 Unspecified	Steel scrap (tinplated)	recycling/reuse	7.73E-02	0	0	0	7.73E-02	1.48E-03
Unspecified nonhazardous waste recycling/reuse 0 1.26E-04 0 0 1.26E-04 2.40E-06 Unspecified sludge landfill 0 3.56E-04 0 0 3.56E-04 6.80E-06 Unspecified sludge recycling/reuse 0 8.46E-01 0 0 8.46E-01 1.62% Unspecified solid waste landfill 2.40E-00 0 0 5.73E-02 1.09E-03 Unspecified solid waste recycling/reuse 1.50E+00 2.11E-01 0 0 1.71E+00 3.27% Unspecified solid waste treatment 4.61E-04 6.44E-04 0 -3.84E-05 1.07E-03 2.04E-05 Unspecified waste recycling/reuse 4.05E-01 1.72E-01 0 0 -9.74E-02 -1.86E-03 Unspecified waste recycling/reuse 0 2.70E-01 0 0 2.70E-01 5.16E-03 Unspecified waste recycling/reuse 0 2.70E-01 0 0 2.70E-01 2.06E-03 Unspec	Sulfur	landfill	7.37E-06	0	0	0	7.37E-06	1.41E-07
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Unspecified nonhazardous waste	recycling/reuse	0	1 26E-04	0	0	1 26E-04	2.40E-06
Unspecified sludge recycling/reuse 0 8,46E-01 0 0 8,46E-01 1,629 Unspecified sludge treatment 0 5,73E-02 0 0 5,73E-02 1,09E-03 Unspecified solid waste landfill 2,40E+00 0 0 -5,10E-01 1,89E+00 3,62% Unspecified solid waste recycling/reuse 1,50E+00 2,11E-01 0 0 1,71E+00 3,27% Unspecified solid waste treatment 0 1,63E+00 0 1,63E+00 3,11% Unspecified solid waste (incinerated) treatment 4,61E-04 6,44E+04 0 -3,84E+05 1,07E+03 2,04E+05 Unspecified waste landfill 0 0 -9,74E+02 -1,86E+03 Used silica gel landfill 0 6,22E+04 0 0 6,22E+04 1,05E+01 Waste acids, unspecified treatment 0 1,05E+01 0,0E+03 2,00E+03 Waste akiali (color filter developer, recycling/reuse 0 3	Unspecified sludge	landfill	0	3 56E-04	0	0	3 56E-04	6 80E-06
Unspecified sludgetreatment05.73E-02005.73E-021.09E-03Unspecified solid wastelandfill $2.40E+00$ 00-5.10E-01 $1.89E+00$ 3.62% Unspecified solid wasterecycling/reuse $1.50E+00$ $2.11E-01$ 00 $1.71E+00$ 3.27% Unspecified solid wastetreatment0 $1.63E+00$ 00 $1.63E+00$ 3.11% Unspecified solid wastetreatment $4.61E-04$ $6.44E-04$ 0 $-3.84E-05$ $1.07E-03$ $2.04E-05$ Unspecified wasterecycling/reuse $4.05E-01$ $1.72E-01$ 0 $-9.83E-03$ $5.67E-01$ $1.08E-02$ Unspecified wasterecycling/reuse $4.05E-01$ $1.72E-01$ 0 $-9.74E-02$ $-9.74E-02$ $-1.86E-03$ Used silica gellandfill0 $6.22E-04$ 00 $6.22E-04$ $1.19E-05$ Waste acid (containing F and detergents)landfill0 $2.70E-01$ 0 $0.270E-01$ $2.00E-03$ Waste aklali (color filter developer,recycling/reuse0 $8.91E-02$ 0 $0.323E-01$ $6.18E-03$ Waste alkali, unspecifiedtreatment0 $1.95E-06$ 0 0 $1.95E-06$ $3.73E-08$ Waste alkali, unspecifiedtreatment0 $1.95E-06$ 0 0 $1.52E-03$ $2.09E-03$ Waste alkali, unspecifiedtreatment0 $1.95E-06$ 0 0 $1.52E-03$ $2.09E-05$ Waste backlight light guide (PMMA)land	Unspecified sludge	recycling/reuse	0	8.46E-01	0	0	8.46E-01	1.62%
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Unspecified sludge	treatment	0	5 73E-02	0	0	5 73E-02	1.09E-03
Unspecified solid wasterecycling/reuse $1.50E+00$ $2.11E-01$ 0 0 $1.71E+00$ 3.27% Unspecified solid wastetreatment 0 $1.63E+00$ 0 0 $1.63E+00$ 3.11% Unspecified solid wastetreatment $4.61E-04$ $6.44E-04$ 0 $-3.84E-05$ $1.07E-03$ $2.04E-05$ Unspecified wasterecycling/reuse $4.05E-01$ $1.72E-01$ 0 $-9.83E-03$ $5.67E-01$ $1.08E-02$ Unspecified wastelandfill 0 0 0 $-9.74E-02$ $-1.86E-03$ Used silica gellandfill 0 0 0 0 $2.70E-01$ 0 Waste acid (containing F and detergents)landfill 0 $2.70E-01$ 0 0 $2.70E-01$ $5.16E-03$ Waste aklali (color filter developer,recycling/reuse 0 $8.91E-02$ 0 0 $8.91E-02$ $1.70E-03$ Waste aklali, unspecifiedtreatment 0 $1.95E-06$ 0 0 $1.95E-06$ $3.23E-01$ $6.18E-03$ Waste aklali, unspecifiedtreatment 0 $1.95E-06$ 0 0 $1.52E-03$ $2.90E-05$ Waste backlight light guide (PMMA)landfill 0 $1.52E-03$ 0 0 $1.52E-03$ $2.90E-01$ Waste oldtreatment 0 $2.03E-01$ 0 0 $2.03E-01$ $5.61E-05$ Waste backlight light guide (PMMA)landfill 0 $1.52E-03$ 0 0 $2.63E-01$ $5.61E-05$ <t< td=""><td>Unspecified solid waste</td><td>landfill</td><td>2.40E+00</td><td>0</td><td>0</td><td>-5.10E-01</td><td>1.89E+00</td><td>3.62%</td></t<>	Unspecified solid waste	landfill	2.40E+00	0	0	-5.10E-01	1.89E+00	3.62%
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Unspecified solid waste	recycling/reuse	1 50E+00	2.11E-01	0	0	1 71E+00	3 2.7%
Inspecified solid waste (incinerated)treatment $4.61E-04$ $6.44E-04$ 0 $-3.84E-05$ $1.07E-03$ $2.04E-05$ Unspecified wasterecycling/reuse $4.05E-01$ $1.72E-01$ 0 $-9.83E-03$ $5.67E-01$ $1.08E-02$ Unspecified wastelandfill 0 0 0 $-9.74E-02$ $-9.74E-02$ $-1.86E-03$ Used silica gellandfill 0 $6.22E-04$ 0 0 $6.22E-04$ $1.19E-05$ Waste acid (containing F and detergents)landfill 0 $2.70E-01$ 0 0 $2.70E-01$ $2.00E-03$ Waste acids, unspecifiedtreatment 0 $1.05E-01$ 0 0 $1.05E-01$ $2.00E-03$ Waste alkali (color filter developer,recycling/reuse 0 $3.23E-01$ 0 0 $3.23E-01$ $6.18E-03$ Waste alkali, unspecifiedtreatment 0 $1.95E-06$ 0 0 $1.95E-06$ $3.73E-08$ Waste backlight casing (PC)landfill 0 $1.46E-05$ 0 0 $1.52E-03$ $2.90E-05$ Waste LCD glasslandfill 0 $2.63E-01$ 0 0 $2.63E-01$ $5.61E-05$ Waste oiltreatment 0 $1.64E-02$ 0 0 $2.93E-03$ $5.61E-05$ Waste oiltreatment 0 $1.64E-02$ 0 0 $2.93E-03$ $5.61E-05$ Waste backlight light guide (PMMA)landfill 0 $2.93E-03$ 0 $2.93E-03$ $5.61E-05$ Waste LCD glass <td>Unspecified solid waste</td> <td>treatment</td> <td>0</td> <td>1.63E+00</td> <td>0</td> <td>0</td> <td>1.63E+00</td> <td>3.11%</td>	Unspecified solid waste	treatment	0	1.63E+00	0	0	1.63E+00	3.11%
Unspecified waste recycling/reuse 4.05E-01 1.72E-01 0 -9.83E-03 5.67E-01 1.08E-02 Unspecified waste landfill 0 0 0 -9.74E-02 -9.74E-02 -1.86E-03 Used silica gel landfill 0 6.22E-04 0 0 6.22E-04 1.19E-05 Waste acid (containing F and detergents) landfill 0 2.70E-01 0 0 2.70E-01 5.16E-03 Waste acids, unspecified treatment 0 1.05E-01 0 0 1.05E-01 2.00E-03 Waste acids, unspecified treatment 0 1.05E-01 0 0 3.23E-01 6.18E-03 Waste alkali, unspecified treatment 0 1.95E-06 0 0 1.95E-06 3.73E-08 Waste alkali, unspecified treatment 0 1.52E-03 0 0 1.52E-03 2.90E-05 Waste backlight light guide (PMMA) landfill 0 1.52E-03 0 0 1.52E-03 2.90E-05 Waste LCD glass landfill 0 2.63E-01 0 0 <	Unspecified solid waste (incinerated)	treatment	4.61E-04	6.44E-04	0	-3.84E-05	1.07E-03	2.04E-05
Unspecified waste landfill 0 0 -9.74E-02 -9.74E-02 -1.86E-03 Used silica gel landfill 0 6.22E-04 0 0 6.22E-04 1.19E-05 Waste acid (containing F and detergents) landfill 0 2.70E-01 0 0 2.70E-01 2.00E-03 Waste acids, unspecified treatment 0 1.05E-01 0 0 8.91E-02 2.00E-03 Waste alkali, color filter developer, recycling/reuse 0 8.91E-02 0 0 8.91E-02 1.70E-03 Waste alkali, unspecified treatment 0 1.95E-06 0 0 1.95E-06 3.73E-08 Waste backlight casing (PC) landfill 0 1.52E-03 0 0 1.52E-03 2.90E-05 Waste backlight light guide (PMMA) landfill 0 2.63E-01 0 0 2.63E-01 5.03E-03 Waste LCD glass recycling/reuse 0 2.93E-03 0 0 2.93E-03 5.61E-05 Waste oil treatment 0 1.64E-02 0 0 2.93E-03	Unspecified waste	recycling/reuse	4.05E-01	1.72E-01	0	-9.83E-03	5.67E-01	1.08E-02
Used silica gel landfill 0 6.22E-04 0 0 6.22E-04 1.19E-05 Waste acid (containing F and detergents) landfill 0 2.70E-01 0 0 2.70E-01 5.16E-03 Waste acids, unspecified treatment 0 1.05E-01 0 0 2.70E-01 5.16E-03 Waste acids, unspecified treatment 0 1.05E-01 0 0 8.91E-02 1.70E-03 Waste alkali (color filter developer, recycling/reuse 0 3.23E-01 0 0 3.23E-01 6.18E-03 Waste alkali, unspecified treatment 0 1.95E-06 0 0 1.95E-06 3.73E-08 Waste backlight casing (PC) landfill 0 1.52E-03 0 0 1.52E-03 2.90E-05 Waste LCD glass landfill 0 2.63E-01 0 0 2.63E-01 5.03E-03 Waste metals, unspecified recycling/reuse 0 7.20E-01 0 0 2.93E-03 5.61E-05 Waste metals, unspecified recycling/reuse 0 2.93E-03 0 0	Unspecified waste	landfill	0	0	0	-9 74E-02	-9 74E-02	-1 86E-03
Waste acid (containing F and detergents) landfill 0 2.70E-01 0 0 2.70E-01 5.16E-03 Waste acids, unspecified treatment 0 1.05E-01 0 0 1.05E-01 2.00E-03 Waste alkali (color filter developer, recycling/reuse 0 8.91E-02 0 0 8.91E-02 1.70E-03 Waste alkali, unspecified recycling/reuse 0 3.23E-01 0 0 3.23E-01 6.18E-03 Waste alkali, unspecified treatment 0 1.95E-06 0 0 1.95E-06 3.73E-08 Waste backlight casing (PC) landfill 0 1.46E-05 0 0 1.52E-03 2.90E-05 Waste LCD glass landfill 0 2.63E-01 0 0 2.93E-03 2.90E-01 5.01E-05 Waste metals, unspecified recycling/reuse 0 7.20E-01 0 0 2.93E-03 5.61E-05 Waste oil treatment 0 1.64E-02 0 0 2.93E-03 5.61E-05 Waste oil treatment 0 1.64E-02 0	Used silica gel	landfill	0	6.22E-04	0	0	6.22E-04	1.19E-05
Waste acids, unspecified treatment 0 1.05E-01 0 0 1.05E-01 2.00E-03 Waste alkali (color filter developer, recycling/reuse 0 8.91E-02 0 0 8.91E-02 1.70E-03 Waste alkali, unspecified recycling/reuse 0 3.23E-01 0 0 3.23E-01 6.18E-03 Waste alkali, unspecified treatment 0 1.95E-06 0 0 1.95E-06 3.73E-08 Waste backlight casing (PC) landfill 0 1.46E-05 0 0 1.52E-03 2.90E-05 Waste LCD glass landfill 0 1.52E-03 0 0 1.52E-03 2.90E-05 Waste netals, unspecified recycling/reuse 0 2.63E-01 0 0 2.63E-01 5.03E-03 Waste LCD glass landfill 0 2.63E-01 0 0 2.93E-03 5.61E-05 Waste metals, unspecified recycling/reuse 0 2.93E-03 0 0 2.93E-03 5.61E-05 Waste oil landfill 3.17E-05 0 0 0 3.17E	Waste acid (containing F and detergents)) landfill	0	2.70E-01	0	0	2.70E-01	5 16E-03
Waste alkali (color filter developer, recycling/reuse 0 8.91E-02 0 0 8.91E-02 1.70E-03 Waste alkali, unspecified recycling/reuse 0 3.23E-01 0 0 3.23E-01 6.18E-03 Waste alkali, unspecified treatment 0 1.95E-06 0 0 1.95E-06 3.73E-08 Waste backlight casing (PC) landfill 0 1.46E-05 0 0 1.52E-03 2.90E-05 Waste backlight light guide (PMMA) landfill 0 1.52E-03 0 0 1.52E-03 2.90E-05 Waste LCD glass landfill 0 2.63E-01 0 0 2.63E-01 5.03E-03 Waste metals, unspecified recycling/reuse 0 7.20E-01 0 0 2.93E-03 5.61E-05 Waste oil treatment 0 1.64E-02 0 0 3.17E-05 6.05E-07 Waste oil landfill 3.17E-05 0 0 3.17E-05 6.05E-07 Waste plastic from LCD modules recycling/reuse 0 7.40E-02 0 7.40E-02 1.41E-03 <td>Waste acids unspecified</td> <td>treatment</td> <td>0</td> <td>1.05E-01</td> <td>0</td> <td>0</td> <td>1.05E-01</td> <td>2.00E-03</td>	Waste acids unspecified	treatment	0	1.05E-01	0	0	1.05E-01	2.00E-03
Waste alkali, unspecified recycling/reuse 0 3.23E-01 0 0 3.23E-01 6.18E-03 Waste alkali, unspecified treatment 0 1.95E-06 0 0 1.95E-06 3.73E-08 Waste backlight casing (PC) landfill 0 1.46E-05 0 0 1.46E-05 2.79E-07 Waste backlight light guide (PMMA) landfill 0 1.52E-03 0 0 1.52E-03 2.90E-05 Waste LCD glass landfill 0 2.63E-01 0 0 2.63E-01 5.03E-03 Waste metals, unspecified recycling/reuse 0 7.20E-01 0 0 2.93E-03 5.61E-05 Waste oil treatment 0 1.64E-02 0 0 1.64E-02 3.14E-04 Waste oil landfill 3.17E-05 0 0 3.17E-05 6.05E-07 Waste plastic from LCD modules recycling/reuse 0 7.40E-02 0 7.40E-02 1.41E-03 Waste plastic from LCD modules treatment 0 4.03E-01 0 0 4.03E-01 7.70E-03 <	Waste alkali (color filter developer.	recycling/reuse	0	8.91E-02	0	0	8.91E-02	1.70E-03
Waste alkali, unspecified treatment 0 1.95E-06 0 0 1.95E-06 3.73E-08 Waste alkali, unspecified treatment 0 1.95E-06 0 0 1.46E-05 2.79E-07 Waste backlight light guide (PMMA) landfill 0 1.52E-03 0 0 1.52E-03 2.90E-05 Waste backlight light guide (PMMA) landfill 0 1.52E-03 0 0 1.52E-03 2.90E-05 Waste LCD glass landfill 0 2.63E-01 0 0 2.63E-01 5.03E-03 Waste metals, unspecified recycling/reuse 0 7.20E-01 0 0 2.93E-03 5.61E-05 Waste oil treatment 0 1.64E-02 0 0 1.64E-02 3.14E-04 Waste oil landfill 3.17E-05 0 0 3.17E-05 6.05E-07 Waste plastic from LCD modules recycling/reuse 0 7.40E-02 0 0 7.40E-02 1.41E-03 Waste plastic from LCD modules treatment 0 4.03E-01 0 0 4.03E-01 7	Waste alkali unspecified	recycling/reuse	0	3 23E-01	0	0	3 23E-01	6 18E-03
Waste hand, imperiated Infill <	Waste alkali, unspecified	treatment	0	1.95E-06	0	0	1.95E-06	3.73E-08
Waste backlight light guide (PMMA) landfill 0 1.52E-03 0 0 1.52E-03 2.90E-05 Waste LCD glass landfill 0 2.63E-01 0 0 2.63E-01 5.03E-03 Waste LCD glass recycling/reuse 0 7.20E-01 0 0 2.93E-03 5.61E-05 Waste metals, unspecified recycling/reuse 0 2.93E-03 0 0 2.93E-03 5.61E-05 Waste oil treatment 0 1.64E-02 0 0 3.14E-04 Waste oil landfill 3.17E-05 0 0 0 3.17E-05 6.05E-07 Waste plastic from LCD modules recycling/reuse 0 7.40E-02 0 0 7.40E-02 1.41E-03 Waste plastic from LCD modules treatment 0 4.03E-01 0 0 4.03E-01 7.70E-03	Waste backlight casing (PC)	landfill	0	1.46E-05	0	0	1.46E-05	2.79E-07
Waste LCD glass landfill 0 2.63E-01 0 0 2.63E-01 5.03E-03 Waste LCD glass recycling/reuse 0 7.20E-01 0 0 7.20E-01 1.38% Waste metals, unspecified recycling/reuse 0 2.93E-03 0 0 2.93E-03 5.61E-05 Waste oil treatment 0 1.64E-02 0 0 3.17E-05 6.05E-07 Waste plastic from LCD modules recycling/reuse 0 7.40E-02 0 0 7.40E-02 1.41E-03 Waste plastic from LCD modules treatment 0 4.03E-01 0 0 4.03E-01 7.70E-03	Waste backlight light guide (PMMA)	landfill	0	1.52E-03	0	0	1.52E-03	2.90E-05
Waste LCD glass recycling/reuse 0 7.20E-01 0 0 7.20E-01 1.38% Waste metals, unspecified recycling/reuse 0 2.93E-03 0 0 2.93E-03 5.61E-05 Waste oil treatment 0 1.64E-02 0 0 1.64E-02 3.14E-04 Waste oil landfill 3.17E-05 0 0 0 3.17E-05 6.05E-07 Waste plastic from LCD modules recycling/reuse 0 7.40E-02 0 0 7.40E-02 1.41E-03 Waste plastic from LCD modules treatment 0 4.03E-01 0 0 4.03E-01 7.70E-03	Waste LCD glass	landfill	0	2.63E-01	0	0	2.63E-01	5.03E-03
Waste metals, unspecified recycling/reuse 0 2.93E-03 0 0 2.93E-03 5.61E-05 Waste oil treatment 0 1.64E-02 0 0 1.64E-02 3.14E-04 Waste oil landfill 3.17E-05 0 0 0 3.17E-05 6.05E-07 Waste plastic from LCD modules recycling/reuse 0 7.40E-02 0 0 7.40E-02 1.41E-03 Waste plastic from LCD modules treatment 0 4.03E-01 0 0 4.03E-01 7.70E-03	Waste LCD glass	recycling/reuse	0	7.20E-01	0	0	7.20E-01	1.38%
Waste oiltreatment01.64E-02001.64E-023.14E-04Waste oillandfill3.17E-050003.17E-056.05E-07Waste plastic from LCD modulesrecycling/reuse07.40E-02007.40E-021.41E-03Waste plastic from LCD modulestreatment04.03E-01004.03E-017.70E-03	Waste metals, unspecified	recycling/reuse	0	2.93E-03	0	0	2.93E-03	5.61E-05
Waste oil landfill 3.17E-05 0 0 0 3.17E-05 6.05E-07 Waste plastic from LCD modules recycling/reuse 0 7.40E-02 0 0 7.40E-02 1.41E-03 Waste plastic from LCD modules treatment 0 4.03E-01 0 0 4.03E-01 7.70E-03	Waste oil	treatment	0	1.64E-02	0	0	1.64E-02	3.14E-04
Waste plastic from LCD modulesrecycling/reuse07.40E-02007.40E-021.41E-03Waste plastic from LCD modulestreatment04.03E-01004.03E-017.70E-03	Waste oil	landfill	3.17E-05	0	0	0	3.17E-05	6.05E-07
Waste plastic from LCD modulestreatment04.03E-01004.03E-017.70E-03	Waste plastic from LCD modules	recvcling/reuse	0	7.40E-02	0	0	7.40E-02	1.41E-03
	Waste plastic from LCD modules	treatment	0	4.03E-01	0	0	4.03E-01	7.70E-03
Waste plastics from LCD monitor landfill 0 4.05E-02 0 0 4.05E-02 7.75E-04	Waste plastics from LCD monitor	landfill	0	4.05E-02	0	0	4.05E-02	7.75E-04
Waste refractory landfill 0 1.13E-04 0 0 1.13E-04 2.16E-06	Waste refractory	landfill	0	1.13E-04	0	0	1.13E-04	2.16E-06
Zinc (elemental) landfill 1.96E-07 0 0 0 1.96E-07 3.75E-09	Zinc (elemental)	landfill	1.96E-07	0	0	0	1.96E-07	3.75E-09
Total solid waste 1.31E+01 1.26E+01 3.11E+01 -4.42E+00 5.23E+01 100.00%	Total solid waste	2	1.31E+01	1.26E+01	3.11E+01	-4.42E+00	5.23E+01	100.00%

Table J-16. LCD solid waste outputs (kg/functional unit)

Material	Disposition	Upstream	Mfg	Use	EOL	Total	% of Total
Highly radioactive waste (Class C)	landfill	2.72e-06	0	0	0	2.72e-06	0.18%
Low-level radioactive waste	landfill	1.28e-04	3.74e-04	6.56e-04	1.24e-07	1.16e-03	78.49%
Radioactive waste (unspecified)	landfill	5.77e-06	0	0	0	5.77e-06	0.39%
Uranium, depleted	landfill	0	1.12e-04	1.97e-04	3.73e-08	3.09e-04	20.93%
Total radioactive was	ste	1.37e-04	4.87e-04	8.52e-04	1.62e-07	1.48e-03	100.00%

 Table J-17. LCD radioactive waste outputs (kg/functional unit)

Metadal	Dimentition	Unstant	Me_	U	FOI	Tetel	9/
Material	Disposition	Upstream	Mig	Use	EOL	Total	% or Fraction of Total
Americium-241 (isotope)	landfill	9.42e+01	0	0	0	9.42e+01	0.0002%
Americium-243 (isotope)	landfill	2.05e+00	0	0	0	2.05e+00	5.11E-08
Antimony-124 (isotope)	surface water	2.12e-02	0	0	0	2.12e-02	5.29E-10
Antimony-124 (isotope)	treatment	0	1.68e+00	2.94e+00	5.59e-04	4.63e+00	1.15E-07
Antimony-125 (isotope)	treatment	0	6.70e+00	1.17e+01	2.23e-03	1.84e+01	4.59E-07
Argon-41 (isotope)	air	0	3.40e+03	5.96e+03	1.13e+00	9.36e+03	0.0233%
Barium-140 (isotope)	treatment	0	1.25e-01	2.18e-01	4.14e-05	3.43e-01	8.55E-09
Bromine-89 (isotope)	air	0	3.93e-04	6.89e-04	1.31e-07	1.08e-03	2.70E-11
Bromine-90 (isotope)	air	0	1.60e-04	2.80e-04	5.32e-08	4.40e-04	1.10E-11
Carbon-14 (isotope)	air	9.35e+00	0	0	0	9.35e+00	2.33E-07
Cesium-134 (isotope)	air	3.58e-04	1.08e-02	1.89e-02	3.59e-06	3.01e-02	7.50E-10
Cesium-134 (isotope)	surface water	1.87e-02	0	0	0	1.87e-02	4.65E-10
Cesium-134 (isotope)	treatment	0	4.49e+00	7.87e+00	1.49e-03	1.24e+01	3.08E-07
Cesium-135 (isotope)	landfill	4.59e+04	0	0	0	4.59e+04	0.1145%
Cesium-136 (isotope)	treatment	0	1.75e-03	3.37e-01	6.40e-05	3.39e-01	8.45E-09
Cesium-137 (isotope)	air	3.58e-04	8.15e-02	1.43e-01	2.71e-05	2.25e-01	5.60E-09
Cesium-137 (isotope)	landfill	1.28e-01	0	0	0	1.28e-01	3.20E-09
Cesium-137 (isotope)	surface water	2.74e-02	0	0	0	2.74e-02	6.82E-10
Cesium-137 (isotope)	treatment	0	6.75e+00	1.18e+01	2.24e-03	1.86e+01	4.63E-07
Chromium-51 (isotope)	air	0	2.13e-01	3.74e-01	7.09e-05	5.87e-01	1.46E-08
Chromium-51 (isotope)	treatment	0	8.10e+00	1.42e+01	2.69e-03	2.23e+01	0.0001%
Cobalt-57 (isotope)	air	0	5.74e-04	1.00e-03	1.91e-07	1.58e-03	3.94E-11
Cobalt-57 (isotope)	treatment	0	1.96e-01	3.43e-01	6.52e-05	5.40e-01	1.35E-08
Cobalt-58 (isotope)	air	3.58e-04	6.66e+00	1.28e+03	2.43e-01	1.29e+03	0.0032%
Cobalt-58 (isotope)	surface water	6.14e-02	0	0	0	6.14e-02	1.53E-09
Cobalt-58 (isotope)	treatment	0	7.98e+01	1.40e+02	2.65e-02	2.20e+02	0.0005%
Cobalt-60 (isotope)	air	3.58e-04	5.51e-02	9.65e-02	1.83e-05	1.52e-01	3.79E-09
Cobalt-60 (isotope)	surface water	3.84e-02	0	0	0	3.84e-02	9.56E-10
Cobalt-80 (isotope)	treatment	0	2.09e+01	3.67e+01	6.96e-03	5.76e+01	0.0001%
Curium-244 (isotope)	landfill	1.91e+02	0	0	0	1.91e+02	0.0005%
Curium-245 (isotope)	landfill	2.13e-02	0	0	0	2.13e-02	5.31E-10
Iodine-129 (isotope)	landfill	3.01e-03	0	0	0	3.01e-03	7.50E-11
Iodine-131 (isotope)	air	2.10e-03	2.57e-01	4.51e-01	8.56e-05	7.11e-01	1.77E-08
Iodine-131 (isotope)	surface water	2.33e-03	0	0	0	2.33e-03	5.80E-11
Iodine-131 (isotope)	treatment	0	3.73e+00	6.54e+00	1.24e-03	1.03e+01	2.56E-07
Iodine-132 (isotope)	air	0	5.23e-02	9.16e-02	1.74e-05	1.44e-01	3.59E-09
Iodine-132 (isotope)	treatment	0	1 41e+00	2.48e+00	4 70e-04	3 89e+00	9 70E-08
Iodine-133 (isotope)	air	4 09e-03	2.39e+02	4 18e+02	7 94e-02	6 57e+02	0.0016%
Iodine-133 (isotope)	treatment	0	1.60e+00	2.80e+00	5 32e-04	4 40e+02	1 10E-07
Iodine-134 (isotope)	air	0	2.71e-01	4 74e-01	9.00e-05	7 45e-01	1 86E-08
Iodine-135 (isotope)	air	0	1 36e-02	2 38e-02	4 52e-06	3 74e-02	9.33E-10
Iodine-135 (isotope)	treatment	0	1.50c 02	2.000 02	3.82e-04	3.16e+00	7.88E-08
Iron-55 (isotope)	treatment	0	1.130+00	3.34e+01	6 34e-03	5.25e+01	0.0001%
Iron-59 (isotope)	treatment	0	9.79e-01	1.72e+00	3.25e-04	2.69e+00	6.72E-08
Krynton-85 (isotope)	air	5.45e+01	5.64e+03	0.88e+03	1.88e+00	1.56e+0.04	0.0388%
Krypton-85M (isotope)	air	0.430+01	2.73e+02	4.79e+02	9.09e-02	7.53e+02	0.0019%
Krypton-85M (isotope)	treatment	0	5.04e+00	8 83e+00	1.68e-03	1 30e+01	3 46F. 07
Krypton-87 (isotope)	air	0	1.02e+0.02	1.78e+0.02	3 38- 02	2.80e+02	0.000704
Krypton-88 (isotope)	air	0	4.77e+02	8 37e+02	1.50e-02	1 31e+02	0.0007%
Lanthanum_140 (isotone)	treatment	0	1 330 01	2 3/2 01	4 430 05	3.67-01	0.005570
Lead-210 (isotope)	air	3 210 01	1.550-01	2.540-01	A.+JC-03	3 21= 01	7.00F.00
Leau-210 (Isotope)	an	5.210-01	0	0	0	5.210-01	/.551-09

Table J-18. LCD radioactivity outputs (Bq/functional unit)

1 abic 9-10.	LCD Tuu	Juctivity	output	<u>5 (Dq/1u</u>	netional	unity	
Material	Disposition	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Manganese-54 (isotope)	air	0	3.03e-03	5.30e-03	1.01e-06	8.33e-03	2.08E-10
Manganese-54 (isotope)	surface water	3.07e-03	0	0	0	3.07e-03	7.65E-11
Manganese-54 (isotope)	treatment	0	5.34e+00	9.35e+00	1.77e-03	1.47e+01	3.66E-07
Molybdenum-99 (isotope)	treatment	0	1.01e+07	1.76e+07	3.35e+03	2.77e+07	69.0936%
Neptunium-237 (isotope)	landfill	2.95e+01	0	0	0	2.95e+01	0.0001%
Niobium-95 (isotope)	air	0	1.20e-04	2.11e-04	4.00e-08	3.31e-04	8.25E-12
Niobium-95 (isotope)	treatment	0	1.37e+00	2.41e+00	4.57e-04	3.78e+00	9.43E-08
Palladium-107 (isotope)	landfill	1.04e-02	0	0	0	1.04e-02	2.58E-10
Plutonium-239 (isotope)	landfill	3.57e+04	0	0	0	3.57e+04	0.0890%
Plutonium-240 (isotope)	landfill	5.08e+04	0	0	0	5.08e+04	0.1267%
Plutonium-241 (isotope)	landfill	1.18e+07	0	0	0	1.18e+07	29.3291%
Plutonium-242 (isotope)	landfill	1.92e+02	0	0	0	1.92e+02	0.0005%
Polonium-210 (isotope)	air	5.52e-01	0	0	0	5.52e-01	1.38E-08
Potassium-40 (isotope)	air	8.57e-02	0	0	0	8.57e-02	2.14E-09
Protactinium-234 (isotope)	air	5.06e-03	0	0	0	5.06e-03	1.26E-10
Protactinium-234 (isotope)	surface water	9.36e-02	0	0	0	9.36e-02	2.33E-09
Radioactive aerosols and halogenes (unspecified)	air	2.81e-02	0	0	0	2.81e-02	7.01E-10
radioactive gas (unspecified)	air	8.98e+02	0	0	0	8.98e+02	0.0022%
Radioactive substance (unspecified)	air	1.29e+00	4.44e+01	0	-1.57e+00	4.41e+01	0.0001%
Radioactive substance (unspecified)	surface water	1.20e-02	4.11e-01		-1.46e-02	4.09e-01	1.02E-08
Radium-222 (isotope)	air	3.26e+00	0	0	0	3.26e+00	8.13E-08
Radium-224 (isotope)	surface water	2.97e-01	0	0	0	2.97e-01	7.40E-09
Radium-226 (isotope)	air	4.31e-01	0	0	0	4.31e-01	1.07E-08
Radium-226 (isotope)	landfill	2.44e+02	0	0	0	2.44e+02	0.0006%
Radium-226 (isotope)	surface water	1.80e+02	0	0	0	1.80e+02	0.0004%
Radium-228 (isotope)	air	4.23e-02	0	0	0	4.23e-02	1.05E-09
Radium-228 (isotope)	surface water	5.93e-01	0	0	0	5.93e-01	1.48E-08
Radon-220 (isotope)	air	9.93e-01	0	0	0	9.93e-01	2.47E-08
Radon-222 (isotope)	air	4.30e+04	0	0	0	4.30e+04	0.1072%
Rubidium-88 (isotope)	air	0	1.12e+00	1.96e+00	3.72e-04	3.08e+00	7.67E-08
Ruthenium-103 (isotope)	treatment	0	1.68e-01	2.94e-01	5.59e-05	4.63e-01	1.15E-08
Samarium-151 (isotope)	landfill	4.25e+01	0	0	0	4.25e+01	0.0001%
Selenium-79 (isotope)	landfill	3.31e-02	0	0	0	3.31e-02	8.25E-10
Silver-110M (isotope)	air	0	3.59e-06	6.29e-06	1.19e-09	9.88e-06	2.46E-13
Silver-110M (isotope)	surface water	9.21e-02	0	0	0	9.21e-02	2.29E-09
Silver-110M (isotope)	treatment	0	1.96e+00	3.43e+00	6.52e-04	5.40e+00	1.35E-07
Sodium-24 (isotope)	treatment	0	2.99e-01	5.23e-01	9.93e-05	8.22e-01	2.05E-08
Strontium-89 (isotope)	treatment	0	3.23e-01	5.65e-01	1.07e-04	8.88e-01	2.21E-08
Strontium-90 (isotope)	landfill	6.86e+03	0	0	0	6.86e+03	0.0171%
Strontium-90 (isotope)	treatment	0	7.59e-02	1.33e-01	2.52e-05	2.09e-01	5.21E-09
Strontium-95 (isotope)	treatment	0	8.36e-01	1.47e+00	2.78e-04	2.30e+00	5.74E-08
Sulfur-136 (isotope)	treatment	0	1.80e-01	3.15e-01	5.99e-05	4.96e-01	1.24E-08
Technetium-99M (isotope)	air	0	1.61e-05	2.83e-05	5.36e-09	4.44e-05	1.11E-12
Technetium-99M (isotope)	landfill	1.40e+00	0	0	0	1.40e+00	3.50E-08
Technetium-99M (isotope)	treatment	0	1.17e-01	2.05e-01	3.89e-05	3.22e-01	8.03E-09
Thorium-228 (isotope)	air	3.57e-02	0	0	0	3.57e-02	8.89E-10
Thorium-228 (isotope)	surface water	1.19e+00	0	0	0	1.19e+00	2.96E-08
Thorium-230 (isotope)	air	7.31e-02	0	0	0	7.31e-02	1.82E-09
Thorium-230 (isotope)	landfill	2.44e+02	0	0	0	2.44e+02	0.0006%
Thorium-230 (isotope)	surface water	8.76e+00	0	0	0	8.76e+00	2.18E-07
Thorium-232 (isotope)	air	2.28e-02	0	0	0	2.28e-02	5.68E-10

Table J-18. LCD radioactivity outputs (Bq/functional unit)

Material	Disposition	Upstream	Mfg	Use	EOL	Total	% or Fraction of Total
Thorium-234 (isotope)	air	5.06e-03	0	0	0	5.06e-03	1.26E-10
Thorium-234 (isotope)	surface water	9.36e-02	0	0	0	9.36e-02	2.33E-09
Tin-113 (isotope)	treatment	0	1.85e-01	3.25e-01	6.16e-05	5.10e-01	1.27E-08
Tin-126 (isotope)	landfill	5.79e-02	0	0	0	5.79e-02	1.44E-09
Tritium-3 (isotope)	air	1.09e+02	7.98e+03	1.40e+04	2.65e+00	2.21e+04	0.0550%
Tritium-3 (isotope)	surface water	1.12e+03	0	0	0	1.12e+03	0.0028%
Tritium-3 (isotope)	treatment	0	5.96e+04	1.04e+05	1.98e+01	1.64e+05	0.4091%
Uranium-234 (isotope)	air	1.28e-01	0	0	0	1.28e-01	3.19E-09
Uranium-234 (isotope)	landfill	1.51e+02	0	0	0	1.51e+02	0.0004%
Uranium-234 (isotope)	surface water	3.09e+00	0	0	0	3.09e+00	7.71E-08
Uranium-235 (isotope)	air	9.55e-04	0	0	0	9.55e-04	2.38E-11
Uranium-235 (isotope)	landfill	2.73e+00	0	0	0	2.73e+00	6.81E-08
Uranium-235 (isotope)	surface water	1.34e-01	0	0	0	1.34e-01	3.35E-09
Uranium-238 (isotope)	air	2.10e-01	0	0	0	2.10e-01	5.23E-09
Uranium-238 (isotope)	landfill	4.23e+01	0	0	0	4.23e+01	0.0001%
Uranium-238 (isotope)	surface water	2.90e+00	0	0	0	2.90e+00	7.23E-08
Xenon-131M (isotope)	air	0	4.60e+02	8.06e+02	1.53e-01	1.27e+03	0.0032%
Xenon-131M (isotope)	treatment	0	6.14e+01	1.07e+02	2.04e-02	1.69e+02	0.0004%
Xenon-133 (isotope)	air	7.63e+02	4.98e+03	1.16e+05	2.21e+01	1.22e+05	0.3045%
Xenon-133 (isotope)	treatment	0	9.43e+03	1.65e+04	3.13e+00	2.60e+04	0.0647%
Xenon-133M (isotope)	air	0	6.59e+04	7.73e+03	1.47e+00	7.36e+04	0.1835%
Xenon-133M (isotope)	treatment	0	7.72e+01	1.35e+02	2.57e-02	2.13e+02	0.0005%
Xenon-135 (isotope)	air	0	2.51e+03	4.39e+03	8.33e-01	6.90e+03	0.0172%
Xenon-135 (isotope)	treatment	0	7.03e+01	1.23e+02	2.34e-02	1.93e+02	0.0005%
Xenon-135M (isotope)	air	0	4.79e+01	8.39e+01	1.59e-02	1.32e+02	0.0003%
Xenon-138 (isotope)	air	0	1.59e+02	2.78e+02	5.28e-02	4.37e+02	0.0011%
Zinc-85 (isotope)	treatment	0	9.00e-02	1.58e-01	2.99e-05	2.48e-01	6.18E-09
Zirconium-93 (isotope)	landfill	1.84e-01	0	0	0	1.84e-01	4.60E-09
Zirconium-95 (isotope)	air	0	3.11e-04	5.44e-04	1.03e-07	8.55e-04	2.13E-11
Total radioactivity outputs	8	1.20e+07	1.02e+07	1.79e+07	3.40e+03	4.01e+07	100%

 Table J-18. LCD radioactivity outputs (Bq/functional unit)

this page intentionally left blank

APPENDIX K

LCIA SUPPORTING TABLES

this page intentionally left blank

APPENDIX K

LCIA SUPPORTING TABLES

Fuel	Heat Value (H) (MJ/L)	Reference	Density (D) (kg/L)	Reference				
Fuel oil # 2 (distillate)	36.739	(1)	0.843	(2)				
Fuel oil # 6 (residual)	38.579	(1)	0.944	(2)				
Fuel oil # 4 (average # 2 & # 6)	37.659	(1)	0.894	(2)				
Liquified natural gas (LNG) ^a	21.185	(3)	0.412	(4)				
Liquified petroleum gas (LPG)	23.276	(1)	0.542	(2)				
Natural gas	0.034	(5)	7.58x 10 ⁻⁴	(6)				

Table K-1. Fuel conversion factors

^a At -260 ° F; 1 ft ³ of liquid methane = 630 ft ³ of gaseous methane.

References:

1. Davis, S.C. 1999. *Transportation Energy Data Book, Edition 19.* 1999. Center for Transportation Analysis, Oak Ridge National Laboratory, ORNL 6958, Appendix B, Table B1. Oak Ridge, Tennessee. September.

2. Energy Information Administration (EIA) 1999. International Energy Annual 1997. U.S. Department of Energy. DOE/EIA-0219 (97), Washington, DC. April.

3. Natural Gas Vehicle Quick Reference Fuel Guide. http:/naturalfuels.com/quick_ref_fuel_guide.htm. Downloaded 8/25/00.

4. Perry, R.H. and Green, D. (Eds.) 1984. Perry's Chemical Engineer's Handbook, 6th Edition, page 9-14. McGraw Hill, Inc., New York, NY.

5. Based on: Wang, M. 1999. The Greenhouse Gases, Regulated Emissions, and energy Use in Transportation (GREET) Model, Version 1.5. Argonne National Laboratory, University of Chicago.

6. Calculated from: Perry, R.H. and D. Green (Eds.) 1984. Perry's Chemical Engineer's Handbook, 6th Edition, page 9-15, Table 9-13, and p. 9-16, Table 9-14. McGraw-Hill, Inc., New York, NY.

Chemical	Synonym	CAS #	GWP ^a
carbon dioxide	CO ₂	124-38-9	1
trifluoromethane	HFC-23	75-46-7	11,700
difluoromethane	HFC-32	75-10-5	650
methyl fluoride	HFC-41	593-53-3	150
1,1,1,2,2,3,4,5,5,5-decafluoropentane	HFC-43-10mee20.8	138495-42-8	1300
pentafluoroethane	HFC-125	354-33-6	2800
1,1,2,2-tetrafluoro-1,2-diiodoethane	HFC-134	359-35-3	1000
1,1,1,2-tetrafluoroethane	HFC-134a	811-97-2	1300
1,1-difluoroethane	HFC-152a	75-37-6	140
1,1,2-trifluoroethane	HFC-143	430-66-0	300
1,1,1-trifluoroethane	HFC-143a	420-46-2	3800
1,1,1,2,3,3,3-heptafluoropropane	HFC-227ea	431-89-0	2900
1,1,1,3,3,3-hexafluoropropane	HFC-236fa	690-39-1	6300
1,1,2,2,3-pentafluoropropane	HFC-245ca	679-86-7	560
sulfur hexafluoride	sulfur hexafluoride	2551-62-4	23,900
carbon tetrafluoride	perfluoromethane	75-73-0	6500
hexafluoroethane/Freon 116	perfluoroethane	76-16-4	9200
octafluoropropane	perfluoropropane	76-19-7	7000
decafluorobutane	perfluorobutane	355-25-9	7000
cyclooctafluorobutane	perfluorocyclobutane	115-25-3	8700
dodecafluoro-pentane	perfluoropentane	678-26-2	7500
tetradecafluorohexane	perfluorohexane	355-42-0	7400
methane	methane	74-82-8	21
nitrous oxide	nitrous oxide	10024-97-2	310

Table K-2.	Global	warming	potentials	(GWP))

^a IPCC's 1995 GWP estimates, 100-year time horizon. Because of the difficulties in calculating indirect effects of CFCs and halons, no indirect values were included.

Source: Houghton et al. 1996.

Chemical	ical Synonym(s) CAS #		Ozone Deple	tion Potential ^e
			ODP ^a	ODP ^b
1,1,1-trichloroethane ^c	methyl chloroform	71-55-56	0.12	0.1
1,2,2-trichloroethane °	methyl chloroform	79-00-5		
CFC-11 °	trichlorofluoromethane	75-69-4	1	1.0
CFC-113 °	trichlorotrifluoroethane	76-13-1	1.07	0.8
CFC-114 °	dichlorotetrafluoroethane	76-14-2	0.8	1.0
CFC-115 °	(mono)chloropentafluoroethane	76-15-3	0.5	0.6
CFC-12 °	dichlorodifluoromethane	75-71-8	1	1.0
CFC-13 °	chlorotrifluoromethane	75-72-9	1	1.0
CFC-111 °	pentachlorofluoroethane	354-56-3		1.0
CFC-112 °	tetrachlorodifluoroethane	76-12-0		1.0
CFC-211 °	heptachlorofluoropropane	N/A		1.0
CFC-212 °	hexachlorodifluoropropane	76564-99-3		1.0
CFC-213 °	pentachlorotrifluoropropane	2354-06-5		1.0
CFC-214 °	tetrachlorotetrafluoropropane	2268-46-4		1.0
CFC-215 °	trichloropentafluoropropane	4259-43-2		1.0
CFC-216 °	dichlorohexafluoropropane	661-97-2		1.0
CFC-217 °	monochloroheptafluoropropane	422-86-6		1.0
CHF2Br	HBFC-22B1; bromodifluoromethane	1511-62-2	0.74	
carbon tetrachloride ^c	tetrachloromethane	56-23-5	1.08	1.1
HALON-1201			1.4	
HALON-1202	difluorodibromomethane	75-61-6	1.25	
HALON-1211 °	bromochlorodifluoromethane	353-59-3	4	3.0
HALON-1301 °	bromotrifluoromethane	75-63-8	16	10.0
HALON-2311		_	0.14	
HALON-2401			0.25	
HALON-2402 °	dibromotetrafluoroethane	124-73-2	7	6.0
HCFC-123 ^d	2,2-dichloro-1,1,1-trifluoroethane	306-83-2	0.02	0.02
HCFC-124 ^d	2-chloro-1,1,1,2-tetrafluorethane	2837-89-0	0.022	0.02
HCFC-141b ^d	1,1-dichloro-1-fluoroethane	1717-00-6	0.11	0.1
HCFC-142b ^d	1-chloro-1,1-difluoroethane	75-68-3	0.065	0.06
HCFC-22 ^d	chlorodifluoromethane	75-45-6	0.055	0.05
HCFC-225ca ^d	3,3-dichloro-1,1,1,2,2-pentafluoropropane	442-56-0	0.025	
HCFC-225cb ^d	1,3-dichloro-1,1,2,2,3-pentafluoropropane	507-55-1	0.033	
bromomethane ^c	methyl bromide	74-83-9	0.6	0.7

 Table K-3. Ozone depletion potentials (ODP)

^a Source: Heijungs *et al.* 1992.
 ^b Listed in Title VI of the 1990 Clean Air Act Amendments (CAAA).

^c Class I substance as listed in Title VI of the 1990 CAAA..

^d Class II substance as listed in Title VI of the 1990 CAAA. (Additional Class III substances listed in the CAAA but not listed here currently have no ODP data.)

^e Weight ratios, compared to CFC-11 = 1.

"-" represents no data.

Chemical/Material	Synonym(s)	CAS #	POCP ^a
1,1,1-trichloroethane	methyl chloroform	71-55-6	0.021
1,2-dichloroethane	ethylene dichloride	107-06-2	0.021
acetone		67-64-1	0.178
acetylene		74-86-2	0.168
alcohols ^b		N/A	0.196
aldehydes ^b		N/A	0.443
benzene		71-43-2	0.189
caprolactam	aminocaproic lactum	105-60-2	0.761
chlorophenols ^b		20-05-3	0.761
crude oil ^b		8002-05-9	0.398
CxHy ^b	hydrocarbons	N/A	0.398
CxHy aliphatic ^b	aliphatic hydrocarbons	N/A	0.398
CxHy aromatic ^b	aromatic hydrocarbons	N/A	0.761
CxHy chloro ^b	chlorinated hydrocarbons	N/A	0.021
dichloromethane	methylene chloride	75-09-2	0.021
diethyl ether	ethyl ether	66-29-7	0.398
diphenyl	biphenyl	92-52-4	0.761
ethanol	ethyl alcohol	64-17-5	0.268
ethene	ethylene	74-85-1	1
ethylene glycol		107-21-1	0.196
ethylene oxide		75-21-8	0.377
formaldehyde		50-00-0	0.421
hexachlorobiphenyl	2,2',4,4',5'5'-hexachloro-1,1-biphenyl	35065-27-1	0.761
hydroxy compounds ^b		N/A	0.377
isopropanol	isopropyl alcohol; 2-propanol	67-63-0	0.196
ketones ^b		N/A	0.326
methane		74-82-8	0.007
methyl ethyl ketone	MEK	78-93-3	0.473
methyl mercaptan		74-93-1	0.377
naphthalene		91-20-3	0.761
non methane VOC ^b		N/A	0.416
PAH ^b	PAC; polycyclic aromatic hydrocarbons	N/A	0.761
pentane		109-66-0	0.408
petrol ^b	gasoline	N/A	0.398
phenol		108-95-2	0.761
phthalic acid anhydride	phthalic anhydride	85-44-9	0.761
propane		74-98-6	0.42
propene	propylene	115-07-1	1.03
propionaldehyde	propanal	123-38-6	0.603

 Table K-4. Photochemical oxidant creation potentials (POCP)

Chemical/Material	Synonym(s)	CAS #	POCP ^a
styrene	vinyl benzene	100-42-5	0.761
terpentine ^b		N/A	0.377
tetrachloromethane	carbon tetrachloride	56-23-5	0.021
toluene		108-88-3	0.563
trichloroethene	trichloroethylene	79-01-6	0.066
vinyl acetate		108-05-4	0.223
vinyl chloride		75-01-4	0.021

Table K-4. Photochemical oxidant creation potentials (POCP)

Source: Heijungs *et al.* 1992. "–" represents no data.

Chemical	Synonym(s)	CAS #	AP ^a
ammonia	NH ₃	7664-41-7	1.88 ^b
hydrochloric acid	HCl	7647-01-0	0.88 ^b
hydrofluoric acid	hydrogen fluoride, HF	7664-39-3	1.6 ^b
nitric oxide	NO	10102-43-9	1.07 ^b
nitrogen dioxide	NO ₂	10102-44-0	0.7 ^b
nitrogen oxides	NO _X	N/A	0.7 ^b
sulfur dioxide	SO_2	7446-09-5	1 ^b
sulfur oxides	SO_X	N/A	1 ^b
sulfur trioxide	SO_3	7446-11-9	0.80 °
nitric acid	HNO ₃	7697-37-2	0.51 °
sulfuric acid	H_2SO_4	7664-93-9	0.65 °
phosphoric acid	H_3O_4P	7664-38-2	0.98 °
hydrogen sulfide	H_2S	7783-06-4	1.88 °

Table K-5. Acidification potentials (AP)

^a Ratios per equal weights, compared to SO₂, SO_x = 1 for emissions to air.
^b Source: Heijungs *et al.* 1992.
^c Source: Hauschild and Wenzel 1997.

Chemical/Parameter (releases to water)	Synonym	CAS #	EP ^a
COD	chemical oxygen demand	N/A	0.022 ^b
ammonia	NH ₃	7664-41-7	0.33 ^b
ammonium ion	$\mathrm{NH_4}^+$	N/A	0.33 ^b
total nitrogen	Ν	N/A	0.42 ^b
phosphate	PO ₄ - ³	N/A	1.0 ^b
total phosphorus	Р	N/A	3.06 ^b
nitrate	NO ₃ -	NA	0.10 °

Table K-6. Eutrophication potential nutrient enrichment chemicals

^a Ratios per equal weights, compared to phosphate = 1.
 ^b Source: Heijungs *et al.* 1992.

[°] Source: Lindfors *et al.* 1995.

NOTE: Eutrophication potentials for releases to air are available but not used in this methodology because partitioning between air and water phases is not considered in this methodology.

Chemical	Svnonvm(s)	CAS #	OTV (mg/m ³)
acetaldehvde	ethanal	75-07-0	0.00027 ª
acetic acid	succinate	64-19-7	0.061 ª
acetonitrile	methyl cvanide	75-05-8	<67 ^b
acetophenone	acetylbenzene	98-86-2	1.5 ^b
acrolein	2-propenal	107-02-8	0.069 ª
acrylic acid	propenoic acid	79-10-7	0.27 ^b
acrylontirile	vinvl cvanide	107-13-1	3.4 ^b
ammonia	NH ₂	7664-41-7	1.0 ^a
aniline		62-53-3	38 ^b
benzene		71-43-2	108 ^b
benzvl chloride	alpha-chlorotoluene	100-44-7	0.21 ^b
1,3-butadiene	butadiene	106-99-0	1 ^b
butanal	butyraldehyde	123-72-8	0.00084 ^a
butanoic acid	butyric acid	107-92-6	0.00035 ^a
1-butanol	butyl alcohol, –	71-36-3	0.077 ª
2-butanone	methyl ethyl ketone	78-93-3	0.68 ^a
n-butylacetate		123-86-4	0.031 ^a
butylacrylate		141-32-2	0.0015 ^a
n-butylpropionate		590-01-2	0.086 ^a
carbon disulfide	CS_2	75-15-0	0.18 ^a
carbon tetrachloride	tetrachloromethane	56-23-5	884 ^b
carbonyl sulfide	carbon oxysulfide	463-58-1	0.25 ^b
chlorine		7782-50-5	0.23 ^b
2-chloroacetophenone	phenyl chloromethyl ketone	532-27-4	0.1 - 0.7 ^b
chlorobenzene		108-90-7	1.0 ^a
chloroform	trichloromethane	67-66-3	650 ^b
m-cresol	3-methylphenol	108-93-4	0.00022 - 0.035 ^b
cumene	isopropylbenzene	98-82-8	0.04 ^b
decaline	veraline (-)form	14727-56-1	2.8 ^a
P-dichlorobenzene	1,4-dichlorobenzene	106-46-7	0.73 ^b
dichloromethane	methylene chloride	75-09-2	640 ^a
diethylamine		109-89-7	0.09 ^a
dimethylamine		124-40-3	0.0014 ^a
1,2-dimethylbenzene	o-xylene	95-47-6	0.78 ^a
1,3-dimethylbenzene	m-xylene	108-38-3	0.54 ª
1,4-dimethylbenzene	p-xylene	106-42-3	0.52 ª
1,1-dimethylhydrazine	N,N-dimethylhydrazine	57-14-7	15 - 65 ^b
dioxane	1.4-diethylene dioxide: 1.4-dioxane	123-91-1	2.9 ^b

 Table K-7. Odor threshold values (OTV)

Chemical	Synonym(s)	CAS #	OTV (mg/m ³)
ethanal	acetaldehyde	75-07-0	0.00027 ^a
ethanethiol	ethylmercaptan	75-08-1	0.000044 ^a
ethanol	ethyl alcohol	64-17-5	0.64 ^a
ethyl acetate		141-78-6	2.1 ª
ethyl acrylate		140-88-5	0.00082 ^a
2-ethyl-5,5-dimethyl-1,3-dioxane			0.0000056 ª
ethyl butyrate		105-54-4	0.00003 ^a
ethylene dichloride	1,2-dichloroethane	107-06-2	25 ^b
ethylene oxide	oxirane	75-21-8	470 ^b
ethylthioethane	diethylsulfide	352-93-2	0.0014 ^a
hydrazine		302-01-2	3.9 - 5.2 ^b
hydrogen sulfide	H_2S	7783-06-4	0.00043 ^a
isopentylacetate	iso-amylacetate	123-92-2	0.075 ^a
isophorone	3,5,5-trimethyl-2-cyclohexenone	78-59-1	1.1 ^b
isopropylbenzene	cumene	98-82-8	0.073 ^a
isopropylpropionate		637-78-5	0.32 ^a
methanal	formaldehyde	50-00-0	0.49 ^a
methanethiol	methyl mercaptan	74-93-1	0.00024 ^a
methanol	methyl alcohol	67-56-1	5.5 ^b
methyl acetate	acetic acid	79-20-9	22 ª
methylamine		74-89-5	0.0012 ^a
3-methylbutanoic acid	isovaleric acid	503-74-2	0.00022 ^a
methyldithiomethane	dimethyldisulfide	624-92-0	0.0015 ^a
methyl hydrazine		60-34-4	1.9 - 5.7 ^b
methyl methacrylate	2-propenoic acid	80-62-6	0.2 ^b
4-methylpentanon-2	methylisobutylketone, MIBK	108-10-1	0.4 ^b
o-cresol	2-methylphenol	95-48-7	0.0018 ^a
<i>m</i> -cresol	3-methylphenol	108-37-4	0.00057 ^a
<i>p</i> -cresol	4-methylphenol	106-44-5	0.00018 ^a
2-methylpropanoic acid	isobutyric acid	79-31-2	0.005 ^a
2-methylpropanol-1	isobutanol	78-73-1	0.035 ª
2-methylpropene	isobutene	115-11-7	15 ª
methyl acrylate	2-propenoic acid, methyl	96-33-3	0.01 ^a
methyl propionate		554-12-1	3.5 ^a
methylthiomethane	dimethylsulfide	75-18-3	0.0003 ^a
naphthalene		91-20-3	0.2 ^b
nitrobenzene		98-95-3	906 ^b
pentanal	valeraldehyde	110-62-3	0.0024 ª

Table K-7. Odor threshold values (OTV)

Chemical	Synonym(s)	CAS #	OTV (mg/m ³)
phenol		108-95-2	0.039 ^a
phosphine		7803-51-2	0.014 - 2.8 ^b
propanal	propionaldehyde	123-38-6	0.0035 ^a
propanoic acid	propionic acid	79-09-4	0.0052 ª
2-propanon	acetone	67-64-1	72 ª
2-propenal	acrolein	107-02-8	0.069 ^a
propionaldehyde	2-propynal	123-38-6	0.003 ^a
propylene dichloride	1,2-dichloropropane	78-87-5	1.2 ^b
propylene oxide	methyloxidrane	75-56-9	24 ^b
pyridine		110-86-1	0.12 ª
quinoline		91-22-5	28 ^b
styrene	vinylbenzene	100-42-5	0.068 ^a
styrene oxide	1-phenyl-1,2-epoxyethane	96-09-3	0.3 ^b
tetrachloroethene	perchloroethene; tetrachloroethylene	127-18-4	8.3 ^a
1,1,2,2-tetrachloroethane	acetylene tetrachloride	79-34-5	50 ^b
terephthaloyldichloride	terephthalic acid dichloride	100-20-9	0.0032 ^a
toluene	methylbenzene	108-88-3	0.6 ^b
trichloroethene	trichloroethylene	79-01-6	3.9 ^a
1,1,1-trichloroethane	methyl chloroform	71-55-6	5.3 ^a
2,4,6-trichlorophenol		88-06-2	0.00016 ^b
triethylamine		121-44-8	1.1 ^b
trimethylamine		75-50-3	0.00026 ^a
1,2,4-trimethylbenzene	pseudocumene	95-63-6	0.14 ^a
1,3,5-trimethylbenzene	mesitylene	108-67-8	0.18 ^a
vinyl acetate		108-05-4	0.4 ^b
glycol ethers		N/A	0.3 ^b
3-methylindole	skatole	83-34-1	0.1 ^b
polycyclic organic matter		N/A	0.074 ^b

 Table K-7. Odor threshold values (OTV)

^a Source: Roos C. 1989 (as cited in Hiejungs et al. 1992).

^b EPA 1992. *Reference Guide to Odor Thresholds for Hazardous Air Pollutants Listed in the Clean Air Act Amendments of 1990.* Washington, DC. EPA/600/R/92/047.

NOTE: When values were available from both sources, the lower value was used.

"-" represents no data.

			Aquatic ecotoxicity							
Cas #	Material	oral SF (mg/kg- day) ⁻¹	inhal SF (mg/kg- day) ⁻¹	WOE (EPA & IARC)(a)	oral NOAEL (b) (mg/kg-day)	inhal NOAEL (b) (mg/m ³)	oral LOAEL (b,c) (mg/kg-day)	inhal LOAEL (b,c) (mg/m ³)	fish LC50 (mg/L)	fish NOAEL (mg/L)
71-55-6	1,1,1-trichloroethane				2.50e+02	1.21e+03	Х	Х	48.0	7.0
76-14-2	1,2-dichlorotetrafluoroethane (CFC 114)				2.73e+02	х	х	Х	XX	xx
106-99-0	1,3-butadiene	Х	1.8	B2,2B	Х	2.80e+03	Х	Х	4.0	1.0
96-48-0	1,4-butanolide	Х	Х	3	Х	Х	100	Х	XX	XX
107-98-2	1-Methoxy-2-propanol				Х	658	Х	Х	XX	XX
872-50-4	1-Methyl-2-pyrrolidinone (NMP)				Х	Х	Х	40	XX	XX
112-34-5	2-(2-butoxyethoxy)ethanol (glycol ether)				х	х		X	XX	xx
124-17-4	2-(2-butoxyethoxy)ethyl acetate				1000	Х	Х	Х	XX	XX
540-84-1	2,2,4-trimethylpentane (Isooctane)								XX	XX
51207-31-9	2,3,7,8-Tetrachlorodibenzo Furan	1.50e+04	1.50e+04	3			24		XX	XX
1746-01-6	2,3,7,8-Tetrachlorodibenzo-p-Dioxin	1.50e+05	1.50e+05	1	9.00e-08	Х	Х	Х	XX	XX
121-14-2	2,4-Dinitrotoluene	0.68	Х	B2	0.2	Х	Х	Х	24.0	6.0
532-27-4	2-Chloroacetophenone				Х	Х	Х	1.0	XX	XX
111-15-9	2-ethoxyl ethylacetate								XX	XX
91-57-6	2-Methylnaphthalene								XX	XX
138526-69-9	3,4,5-trifluorobromobenzene								XX	XX
348-61-8	3,4-difluorobromobenzene								XX	XX
56-49-5	3-Methylcholanthrene				Х	Х	2.86	Х	XX	XX
82832-73-3 (d)	4-(4-propylcyclohexyl)cyclohexanone								XX	XX
106-41-2	4-bromophenol								XX	XX
123-07-9	4-ethylphenol								XX	XX
14938-35-3	4-pentylphenol								XX	XX
70-70-2	4-propionylphenol								XX	XX
3697-24-3	5-Methyl chrysene (category: PAH)	X	X	2B					XX	XX
83-32-9	Acenaphthene (category: PAH)				175	X	350	X	XX	XX
208-96-8	Acenaphthylene (category: PAH)	Х	X	D					XX	XX
75-07-0	Acetaldehyde	Y	7.70e-03	2B	125	300	Х	Х	34.0	9.0

Table K-8. Chemicals in the CDP inventory classified as potentially toxic

				Aquatic ecotoxicity						
Cas #	Material	oral SF (mg/kg- day) ⁻¹	inhal SF (mg/kg- day) ⁻¹	WOE (EPA & IARC)(a)	oral NOAEL (b) (mg/kg-day)	inhal NOAEL (b) (mg/m ³)	oral LOAEL (b,c) (mg/kg-day)	inhal LOAEL (b,c) (mg/m ³)	fish LC50 (mg/L)	fish NOAEL (mg/L)
64-19-7	Acetic acid				195	Х	Х	Х	XX	XX
67-64-1	Acetone	Х	Х	D	100	Х	Х	Х	720	180
98-86-2	Acetophenone				423	Х	Х	Х	XX	XX
74-86-2	Acetylene								XX	XX
107-02-8	Acrolein	Х	Х	С,3					XX	XX
No CAS #	Aluminium (Al3+)								3.6	0.36
7429-90-5	Aluminum (Al)	Х	Х	SAR0	60	Х	Х	Х	XX	XX
1344-28-1	Aluminum oxide								XX	XX
7664-41-7	Ammonia				34	40	Х	Х	2.0	9.00e-02
1341-49-7	Ammonium bifluoride				5.10e-02	Х	Х	Х	XX	XX
7789-09-5	Ammonium Dichromate	Х	Х	Al					XX	XX
12125-01-8	Ammonium Fluoride								XX	XX
1336-21-6	Ammonium hydroxide								XX	XX
1113-38-8	Ammonium Oxalate								XX	XX
6009-70-7	Ammonium Oxalate Monohydrate								XX	XX
628-63-7	Amyl Acetate (mixed isomers)								XX	XX
120-12-7	Anthracene (category: PAH)	Х	Х	SAR1	1000	Х	Х	Х	0.01	
7440-36-0	Antimony (Sb)				Х	Х	0.35	Х	14.4	1.6
7440-38-2	Arsenic (As)	1.5	50	А	8.00e-04	Х	Х	Х	14.4	2.1
"20-01-9"	Arsenic compounds [Arsenic (As3+, As5+)]	1.5	х	A,1	8.00e-04	х	х	х	32.0	2.0
7440-39-3	Barium (Ba)				0.21	Х	Х	Х	580	50.0
513-77-9	Barium carbonate	Х	Х	D	0.21	Х	Х	Х	XX	XX
"20-02-0"	Barium compounds [Barium (Ba++)]	Х	Х	D	0.21	Х	Х	Х	200	10.0
7727-43-7	Barium Sulfate	Х	Х	D	0.21	Х	Х	Х		
100-52-7	Benzaldehyde				143	Х	Х	Х	27.0	XX
71-43-2	Benzene	0.055	0.029	A,1	1.0	1.15	10	98	19.0	4.0
56-55-3	Benzo(a)anthracene (category: PAH)	0.73	0.31	B2					XX	XX
50-32-8	Benzo(a)pyrene	7.3	3.1	B2,2A					XX	XX

Table K-8. Chemicals in the CDP inventory classified as potentially toxic

			Aquatic ecotoxicity							
Cas #	Material	oral SF (mg/kg- day) ⁻¹	inhal SF (mg/kg- day) ⁻¹	WOE (EPA & IARC)(a)	oral NOAEL (b) (mg/kg-day)	inhal NOAEL (b) (mg/m ³)	oral LOAEL (b,c) (mg/kg-day)	inhal LOAEL (b,c) (mg/m ³)	fish LC50 (mg/L)	fish NOAEL (mg/L)
56832-73-6	Benzo(b,j,k)fluoranthene (category: PAH)	х	х	B2					XX	XX
191-24-2	Benzo(g,h,i)perylene (category: PAH)	х	х	D					XX	XX
205-99-2	Benzo[b]fluoranthene	0.73	0.31	B2					XX	XX
100-44-7	Benzyl chloride	0.17	Х	B2,3					XX	XX
7440-41-7	Beryllium (Be)	4.3	8.4	Х	Х	Х	Х	5.50e-04	XX	XX
92-52-4	Biphenyl (category: PAH)	Х	Х	SAR0	50	Х	Х	Х	2.0	0.12
68611-71-2 (d)	Blue phosphor (ZnS)								XX	XX
1314-98-3	Blue phosphor (ZnS:Ag:Al)								XX	XX
1303-96-4	borax								XX	XX
11113-50-1	boric acid				67	Х	62.5	Х		
No CAS #	Boron (B III)				8.8	Х	Х	Х	113	27.0
7440-42-8	Boron (B)				8.8	Х	Х	Х	113	27.0
7726-95-6	Bromine								XX	XX
75-25-2	Bromoform	7.90e-03	3.90e-03	B2	17.9	Х	Х	Х	XX	XX
74-83-9	Bromomethane [Methyl bromide]	Х	Х	C,3	0.4	4.3	Х	Х	11.0	3.0
7440-43-9	Cadmium (Cd)	Х	6.1	B1,1	Х	Х	4.00e-02	2.20e-02	0.001	0.001
"20-04-2"	Cadmium cmpds (as CdCl2) [Cadmium (Cd++)]	х	х	B1,2A	5.00e-03	х	х	Х	0.1	
75-15-0	Carbon disulfide				Х	10	х	Х	694	174
630-08-0	Carbon monoxide (CO)				Х	114.5	х	55	XX	XX
56-23-5	Carbon tetrachloride	1.30e-01	5.30e-02	B2,2B	1	34.3	Х	Х	41.0	5.0
No CAS #	Cerium (Ce++)									
1306-38-3	Cerium oxide				Х	Х	х	5.0	XX	XX
No CAS #	Cesium (Cs++)									
75-72-9	CFC 13								XX	XX
7782-50-5	Chlorine (Cl2)				14.0	Х	Х	Х	0.34	0.02
1341-24-8	Chloroacetophenone								XX	XX
108-90-7	Chlorobenzene	Х	Х	SAR0	12.5	377	Х	Х	17.0	2.0

 Table K-8.
 Chemicals in the CDP inventory classified as potentially toxic

			Aquatic ecotoxicity							
Cas #	Material	oral SF (mg/kg- day) ⁻¹	inhal SF (mg/kg- day) ⁻¹	WOE (EPA & IARC)(a)	oral NOAEL (b) (mg/kg-day)	inhal NOAEL (b) (mg/m ³)	oral LOAEL (b,c) (mg/kg-day)	inhal LOAEL (b,c) (mg/m ³)	fish LC50 (mg/L)	fish NOAEL (mg/L)
67-66-3	Chloroform	6.10e-03	8.10e-02	B2,2B	Х	Х	12.9	Х	71.0	18.0
16065-83-1	Chromium (Cr III)	Х	Х	D	1468	Х	Х	Х	3.3	0.33
7440-47-3	Chromium (Cr)	Х	Х	1					52.0	5.2
1333-82-0	Chromium oxide (chromium trioxide)	Х	Х	D	1468	Х	Х	Х	XX	XX
18540-29-9	Chromium, hexavalent	Х	41	A,1	2.5	Х	Х	Х	22.6	2.23
218-01-9	Chrysene (category: PAH)	7.30e-03	3.10e-03	Х					XX	XX
No CAS #	Cobalt (Co I, Co II, Co III)									
7440-48-4	Cobalt (Co)								XX	XX
7440-50-8	Copper (Cu)	Х	Х	D	5.30e-01	Х	Х	Х	1.40e-02	4.00e-03
No CAS #	Copper (Cu+, Cu++)				5.30e-01	Х	Х	Х	1.40e-02	4.00e-03
9065-82-1	cresol-formaldehyde resins								XX	XX
98-82-8	Cumene	Х	Х	SAR0	154	537	Х	Х	6.0	0.49
80-15-9	Cumene hydroperoxide	Х	Х	SAR1	Х	31	Х	Х	62.0	16.0
57-12-5	Cyanide (CN)	Х	Х	D	10.8	Х	Х	Х	56.0	5.7
110-82-7	Cyclohexane	Х	Х	SAR0	Х	1500	Х	Х	5.0	0.39
117-81-7	Di(2-ethylhexyl)phthalate [Bis(2-ethylhexyl)phthalate]	х	х	B2,2B	50	50	х	Х	1.0	0.08
53-70-3	Dibenzo(a,h)anthracene	7.3	3.1	B2					XX	XX
25321-22-6	Dichlorobenzene (mixed isomers)	Х	Х	SAR0	Х	610.4	Х	Х	1.0	0.05
75-71-8	Dichlorodifluoromethane (CFC 12)				15	Х	Х	Х	XX	XX
75-09-2	Dichloromethane (Methylene chloride)	7.50e-03	1.65e-03	B2,2B	155	796	х	Х	330	83.0
68334-30-5	Diesel fuel	Х	Х	С					XX	XX
60-29-7	Diethyl ether (Ethyl ether)				500	Х	Х	Х	XX	XX
111-46-6	Diethylene Glycol				1250	Х	Х	Х	XX	XX
68-12-2	Dimethyl formamide				Х	Х	Х	7.9	XX	XX
77-78-1	Dimethyl sulfate	Х	Х	B1,2A					XX	XX
57-97-6	Dimethylbenzanthracene				Х	Х	Х	1.40e-02	XX	XX
67-68-5	di-methyl-sulfoxide				Х	Х	1.0	Х	XX	XX
122-62-3	Dioctyl Sebacate				200	х	Х	x	XX	XX

Table K-8. Chemicals in the CDP inventory classified as potentially toxic

			Aquatic ecotoxicity							
Cas #	Material	oral SF (mg/kg- day) ⁻¹	inhal SF (mg/kg- day) ⁻¹	WOE (EPA & IARC)(a)	oral NOAEL (b) (mg/kg-day)	inhal NOAEL (b) (mg/m ³)	oral LOAEL (b,c) (mg/kg-day)	inhal LOAEL (b,c) (mg/m ³)	fish LC50 (mg/L)	fish NOAEL (mg/L)
60-00-4	Edetic Acid (EDTA)								473	240
74-84-0	Ethane								XX	XX
75-08-1	Ethanethiol [Mercaptans]								XX	XX
141-43-5	ethanol amine				320	Х	Х	12.7	XX	XX
75-00-3	Ethyl chloride	Х	Х	3	Х	3600	Х	Х	16.0	4.0
100-41-4	Ethylbenzene	Х	Х	SAR0	136	2370	Х	Х	11.0	1.0
74-85-1	Ethylene	Х	Х	SAR0	Х	11600	Х	Х	14.0	3.0
106-93-4	Ethylene dibromide	85	7.60e-01	B2					XX	XX
107-06-2	Ethylene dichloride	9.10e-02	9.10e-02	B2,2B	18	221	Х	Х	136	34.0
unknown	Etoxy Naphtol Sulphonic Acid (ENSA)								XX	xx
No CAS #	Ferromanganese (Fe, Mn, C)								XX	XX
206-44-0	Fluoranthene (category: PAH)	Х	Х	D	125	Х	Х	Х	XX	XX
86-73-7	Fluorene (category: PAH)	Х	Х	D	125	Х	Х	Х	XX	XX
16984-48-8	Fluoride									
No CAS #	Fluorides (F-)				6.00e-02	Х	Х	Х		
7782-41-4	Fluorine (F2)				6.00e-02	Х	Х	Х	XX	XX
9002-84-0	Fluorocarbon resin [Tetrafluoroethylene (C2F4)]	х	х	3					XX	xx
50-00-0	Formaldehyde (CH2O)	Х	4.50e-02	B1,2A	15	0.6	Х	Х	24.0	6.0
No CAS #	Fuel Oil #2 (distillate and diesel)								XX	XX
No CAS #	Fuel Oil #4 (distillate and residual)								XX	XX
No CAS #	Fuel Oil #6 (residual)								XX	XX
111-76-2	glycol ethers [2-butoxy ethanol]	Х	Х	C,3	203	121	х	Х	1,490	373
unknown	Green Phosphor (ZnS.Cu.Al)								XX	XX
68611-68-7 (d)	Green phosphors (ZnS)								XX	XX
75-63-8	Halon 1301								XX	XX
75-45-6	HCFC 22				X	5,260	X	X	XX	XX

 Table K-8. Chemicals in the CDP inventory classified as potentially toxic

		Chronic								Aquatic ecotoxicity	
Cas #	Material	oral SF (mg/kg- day) ⁻¹	inhal SF (mg/kg- day) ⁻¹	WOE (EPA & IARC)(a)	oral NOAEL (b) (mg/kg-day)	inhal NOAEL (b) (mg/m ³)	oral LOAEL (b,c) (mg/kg-day)	inhal LOAEL (b,c) (mg/m ³)	fish LC50 (mg/L)	fish NOAEL (mg/L)	
422-56-0	HCFC-225ca								XX	XX	
507-55-1	HCFC-225cb								XX	XX	
142-82-5	Heptane (n-Heptane)	Х	Х	D	1,000	Х	Х	1,630	XX	XX	
67-72-1	Hexachloroethane	1.40e-02	1.40e-02	С,3	1.0	Х	Х	Х	1.0	0.35	
999-97-3	Hexamethyldisilazane (HMDS)								XX	XX	
110-54-3	Hexane				Х	Х	Х	73	2.5	0.25	
354-33-6	HFC 125				Х	2.45e+05	Х	Х	XX	XX	
302-01-2	Hydrazine	3	17	B2					4.83	0.48	
7647-01-0	hydrochloric acid	Х	Х	3	Х	15	Х	Х	19.0	0.95	
7664-39-3	Hydrofluoric acid (hydrogen fluoride)								265	13	
74-90-8	Hydrogen Cyanide	Х	Х	SAR0	10.8	Х	30	7.07	1,385	346	
7722-84-1	Hydrogen Peroxide	Х	Х	3					XX	XX	
7783-06-4	Hydrogen Sulfide				3.1	Х	Х	15	XX	XX	
7790-92-3	Hypochlorous Acid										
193-39-5	Indeno(1,2,3-cd)pyrene (category: PAH)	7.30e-01	3.10e-01	B2					XX	xx	
50926-11-9	Indium tin oxide (ITO)								XX	XX	
123-92-2	Isopentyl acetate (Amyl Acetate)								XX	XX	
78-59-1	Isophorone	9.50e-04	Х	С	150	Х	Х	Х	XX	XX	
67-63-0	Isopropyl alcohol	Х	Х	1	230	268.3	Х	Х	8,623	2,156	
637-78-5	Isopropylpropionate								XX	XX	
7439-91-0	Lanthanum (La)								XX	XX	
7439-92-1	Lead (Pb)	Х	Х	B2,2B					31.5	0.004	
"20-11-1"	Lead compounds (as PbCl2) [Lead (Pb++, Pb4+)]	х	х	B2,2B					5.0	0.26	
1317-36-8	Lead oxide	X	X	B2					XX	XX	
7446-14-2	Lead sulfate cake	X	Х	B2					60.8	6.08	
NA	Liquified petroleum gas (LPG)								2600	260	
NA	Lithium Salts (Lithine)										
NA	LNG (Liquified natural gas)								XX	XX	

Table K-8. Chemicals in the CDP inventory classified as potentially toxic

		Chronic								Aquatic ecotoxicity	
Cas #	Material	oral SF (mg/kg- day) ⁻¹	inhal SF (mg/kg- day) ⁻¹	WOE (EPA & IARC)(a)	oral NOAEL (b) (mg/kg-day)	inhal NOAEL (b) (mg/m ³)	oral LOAEL (b,c) (mg/kg-day)	inhal LOAEL (b,c) (mg/m ³)	fish LC50 (mg/L)	fish NOAEL (mg/L)	
7439-96-5	Manganese	Х	Х	D	0.14	Х	Х	0.15			
"21-12-2"	Manganese cmpds (as MnCl2) [Manganese (Mn II, Mn IV, Mn VII)]	х	х	D	0.14	х	Х	0.15	150.0	8.0	
7439-97-6	Mercury (Hg)	Х	Х	D,3	Х	6.00e-03	Х	9.00e-03	0.155	0.005	
no CAS#	Mercury cmpds (as HgCl2) [Mercury (Hg+, Hg++)]	х	Х	С	Х	х	0.226	Х	0.155	0.005	
74-82-8	Methane (natural gas)								XX	XX	
67-56-1	Methanol	Х	Х	SAR0	500	130	Х	Х	29,400	7,350	
74-87-3	Methyl chloride	1.30e-02	6.30e-03	С,3	Х	1138.4	Х	1550	550	138	
78-93-3	Methyl ethyl ketone	Х	Х	D	125	8047	Х	Х	3,220	805	
60-34-4	Methyl hydrazine	3	17.2	A3					XX	XX	
80-62-6	Methyl methacrylate	Х	Х	SAR0	7.5	111.7	х	Х	259	65	
1634-04-4	Methyl tert butyl ether	Х	Х	SAR0	100	2880	х	Х	786	197	
7439-98-7	Molybdenum (Mo)				Х	Х	0.14	Х	157	0.125	
no CAS#	Molybdenum cmpds [Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI)]								157	0.125	
7803-62-5	Monosilane Gas								XX	XX	
110-91-8	Morpholine	Х	Х	3	Х	Х	Х	36			
91-20-3	Naphthalene	Х	Х	С	71	Х	Х	9.3	6.0	0.59	
79-15-2	N-bromoacetamide (NBA)								XX	XX	
123-86-4	N-butyl acetate [Butyl acetate]				Х	Х	Х	210	XX	XX	
7440-00-8	Neodymium (Nd)								XX	XX	
7440-02-0	Nickel (Ni)	Х	Х	А	5	Х	Х	Х	2.48	0.09	
"20-14-4"	Nickel cmpds (as NiCl2) [Nickel (Ni++, Ni3+)]	х	х	A,1					27	1.0	
14797-55-8	Nitrates				1.6	Х	Х	Х	2,213	213	
7697-37-2	Nitric Acid								26	1.0	
14797-65-0	Nitrites (NO2-)				1.6	Х	Х	Х	225	1.0	
10102-44-0	Nitrogen Dioxide								196	19.6	

Table K-8. Chemicals in the CDP inventory classified as potentially toxic
		Chronic Aquatic ecoto						ecotoxicity		
Cas #	Material	oral SF (mg/kg- day) ⁻¹	inhal SF (mg/kg- day) ⁻¹	WOE (EPA & IARC)(a)	oral NOAEL (b) (mg/kg-day)	inhal NOAEL (b) (mg/m ³)	oral LOAEL (b,c) (mg/kg-day)	inhal LOAEL (b,c) (mg/m ³)	fish LC50 (mg/L)	fish NOAEL (mg/L)
7783-54-2	Nitrogen fluoride (nitrogen trifluoride)								xx	xx
no CAS#	Nitrogen Oxides (NOx)								XX	XX
10024-97-2	Nitrous oxide								XX	XX
10043-35-3	orthoboric acid				67	Х	62.5	Х	XX	XX
95-47-6	o-xylene	Х	Х	D	179	Х	Х	Х	16.0	2.0
608-93-5	Pentachlorobenzene	Х	Х	D	Х	Х	8.3	Х	XX	XX
87-86-5	Pentachlorophenol	Х	Х	B2	3	Х	Х	Х	XX	XX
109-66-0	Pentane	Х	Х	D					XX	XX
7601-90-3	Perchloric acid								XX	XX
76-16-4	Perfluoroethane (Hexafluorocarbon)				Х	Х	Х	1.17e+06	XX	XX
75-73-0	Perfluoromethane (CF4)								XX	XX
10450-60-9	Periodic Acid								XX	XX
No CAS #	Petroleum								XX	XX
85-01-8	Phenanthrene (category: PAH)	Х	Х	D					XX	XX
108-95-2	Phenol	Х	Х	D,3	60	Х	Х	Х	34.0	8.0
98-67-9	Phenolsulphonic Acid								XX	XX
57583-54-7 (d)	Phosphate ester, plastic components				1300	Х	Х	Х	XX	XX
7803-51-2	Phosphine gas				0.026	0.25	Х	Х	XX	XX
7664-38-2	phosphoric acid				Х	50	Х	180	70.0	4.0
7723-14-0	Phosphorus	Х	Х	D	1.50e-02	Х	Х	Х	0.02	
1314-56-3	Phosphorus Pentoxide									
NA	PM [particulates, total]								XX	XX
NA	PM-10 [Particulates < 10 microns]								XX	XX
1336-36-3	Polychloriniated biphenyl (PCB)	Х	Х	B2,2A	7.00e-03	Х	Х	Х	3.0	0.14
9016-45-9	Polyethylene mono(nonylphenyl)ether glycol [Tergitol NP-33 (glycol ether)]				1000	х	67.5	х	XX	xx
9002-89-5	Polyvinyl alcohol	Х	Х	3					XX	XX
9003-39-8	Polyvinyl Pyrrolidone (PVP)	Х	Х	3	550	Х	5500	Х	XX	XX

Table K-8. Chemicals in the CDP inventory classified as potentially toxic

		Chronic						Aquatic ecotoxicity		
Cas #	Material	oral SF (mg/kg- day) ⁻¹	inhal SF (mg/kg- day) ⁻¹	WOE (EPA & IARC)(a)	oral NOAEL (b) (mg/kg-day)	inhal NOAEL (b) (mg/m ³)	oral LOAEL (b,c) (mg/kg-day)	inhal LOAEL (b,c) (mg/m ³)	fish LC50 (mg/L)	fish NOAEL (mg/L)
8486041	PPE [Polyphenylene ether]								XX	XX
123-38-6	Propionaldehyde	Х	Х	SAR3	Х	200	Х	Х	44.0	11.0
115-07-1	Propylene	Х	Х	SAR0	Х	9375	Х	Х	5.0	1.0
108-32-7	Propylene carbonate								XX	XX
57-55-6	propylene glycol				Х	170	Х	Х	XX	XX
108-65-6	propylene glycol monomethyl ether acetate [1-Methoxy-2-propyl Acetate (glycol ether)]								XX	xx
129-00-0	Pyrene (category: PAH)	Х	Х	D	75	Х	Х	Х	XX	XX
68784-83-8 (d)	Red phosphors								XX	XX
No CAS#	Red phosphors (Y ₂ O ₂ S.Eu)								XX	XX
No CAS #	Rubidium (Rb+)									
7440-20-2	Scandium (Sc)								XX	XX
7782-49-2	Selenium (Se)	Х	Х	D	1.50e-02	Х	Х	Х	XX	XX
7440-21-3	Silicon (Si)								XX	XX
7440-22-4	Silver	Х	Х	D	х	Х	1.40e-02	Х	4.00e-03	0.001
no CAS#	Silver compounds [Silver (Ag+)]	Х	Х	D	Х	Х	1.40e-02	Х	12.0	0.001
10588-01-9	Sodium Dichromate	Х	Х	3	Х	Х	0.18	0.25	XX	XX
2151247	Sodium Dichromate Dihydrate (VI)								XX	XX
13472-35-0	sodium dihydrogen phosphate dihydrate								XX	xx
7681-52-9	Sodium Hypochlorite	Х	Х	3	2.1	Х	Х	Х	XX	XX
7681-57-4	Sodium Metabisulfite	Х	Х	3					XX	XX
7775-27-1	Sodium Persulfate								XX	XX
No CAS #	Strontium (Sr II)				190	Х	Х	Х		
7440-24-6	Strontium (Sr)				190	Х	Х	Х	XX	XX
1633-05-2	Strontium carbonate				190	Х	Х	Х	XX	XX
100-42-5	Styrene	X	Х	C,2B	100	565	Х	Х	4.0	0.44
7446-09-5	Sulfur dioxide	X	X	3	X	0.104	X	Х	XX	XX
2551-62-4	sulfur hexafluoride								XX	XX

 Table K-8.
 Chemicals in the CDP inventory classified as potentially toxic

		Chronic Aquatic eco						ecotoxicity		
Cas #	Material	oral SF (mg/kg- day) ⁻¹	inhal SF (mg/kg- day) ⁻¹	WOE (EPA & IARC)(a)	oral NOAEL (b) (mg/kg-day)	inhal NOAEL (b) (mg/m ³)	oral LOAEL (b,c) (mg/kg-day)	inhal LOAEL (b,c) (mg/m ³)	fish LC50 (mg/L)	fish NOAEL (mg/L)
no CAS#	sulfur oxides (SOx)								XX	XX
7664-93-9	sulfuric acid	Х	Х	1	Х	0.1	Х	Х	31.0	2.0
10124-29-5	Sulfuric acid, aluminum salt				Х	Х	154	Х	XX	XX
127-18-4	tetrachloroethylene	5.20e-02	2.00e-03	B2,2B	14	740.2	Х	Х	17.0	2.0
109-99-9	Tetrahydrofuran (THF)				782	0.2	Х	Х	XX	XX
75-59-2	Tetramethyl ammonium hydroxide (TMAH)								XX	xx
7440-28-0	Thallium (TI)								XX	XX
7440-29-1	Thorium (Th)	Х	Х	А					XX	XX
7440-31-5	Tin (Sn)								626.0	62.6
No CAS #	Tin (Sn++, Sn4+)								626.0	62.6
7440-32-6	Titanium	Х	Х	С	Х	0.8	1146	Х		
7550-45-0	Titanium tetrachloride				Х	9.00e-03	Х	Х	25.0	1.0
108-88-3	Toluene	Х	Х	D,3	100	411.1	Х	Х	34.0	4.0
1025-15-6	triallyl isocyanurate								XX	XX
79-01-6	trichloroethylene (TCE)	1.10e-02	6.00e-03	B2,3	24	586.6	Х	Х	44.0	8.0
75-69-4	Trichlorofluoromethane (CFC 11)				Х	Х	349	Х	XX	XX
1330-78-5	Tricresyl phosphate								XX	XX
112-27-6	Triethylene Glycol				Х	Х	1200	Х	8.81e+04	8,810
115-86-6	Triphenyl phosphate								XX	XX
7440-33-7	Tungsten (W)								XX	XX
1344-59-8	U ₃ O ₈ (yellowcake)								XX	XX
7440-61-1	Uranium (U)	Х	Х	A1	0.2	Х	Х	Х	XX	XX
7440-62-2	Vanadium (V)				3.00e-03	Х	Х	Х	XX	XX
No CAS #	Vanadium (V3+, V5+)									
108-05-4	Vinyl acetate	Х	Х	SAR0	100	176	Х	Х	100	25.0
75-01-4	Vinyl chloride	1.4	3.08e-02	A,1	Х	6.98e+04	Х	Х	143	36.0
1330-20-7	Xylene (C24H30) [mixed isomers]	X	Х	D	179	Х	Х	Х	13	1.0
7440-66-6	Zinc (Zn)	Х	Х	D	0.9	Х	1	Х	9.00e-02	0.036
No CAS #	Zinc (Zn++)								17.0	

Table K-8. Chemicals in the CDP inventory classified as potentially toxic

	rusto it of chemicult in the CD1 intentity eluspited us petentiuny toxic										
			Chronic							Aquatic ecotoxicity	
Cas #	Material	oral SF (mg/kg- day) ⁻¹	inhal SF (mg/kg- day) ⁻¹	WOE (EPA & IARC)(a)	oral NOAEL (b) (mg/kg-day)	inhal NOAEL (b)	oral LOAEL (b,c)	inhal LOAEL (b,c)	fish LC50 (mg/L)	fish NOAEL (mg/L)	
						(mg/m^3)	(mg/kg-day)	(mg/m^3)			
14940-68-2	Zircon sand [Zircon (Zr)]								XX	XX	
7440-67-7	Zirconium (Zr)				3,494	Х	Х	Х	XX	XX	

Table K-8. Chemicals in the CDP inventory classified as potentially toxic

Key:

(a)=See Table 3-3 in Section 3.1.2.12 for a description of WOE classifications.

(b)=only lowest value of the NOAEL (or LOAEL/10) is used to calculate chronic, non-cancer effects

(c)=LOAEL only needed if no NOAEL found

(d)=CAS # was provided by a company, but could not be confirmed.

XX=aquatic toxicity data not needed because there are no waterborne releases of this chemical in the CDP LCIs.

X=data not needed because other data are provided to calculate impact score (e.g., LOAEL not needed if NOAEL provided, and WOE used if SF not available).

SAR0=not a probable carcinogen based on structure-activity relationship (SAR) evaluation.

SAR1=possible carcinogen based on SAR evaluation.

- - =no data available, defaulted to mean hazard value (see Section 3.1.2.12 for an explanation of hazard values).

Sources:

Oral and inhalation slope factors (SF): Integrated Risk Information System (IRIS) or Health Effects Assessment Summary Tables (HEAST) (EPA, 1994) as cited in Risk Assessment Information System (RAIS): http://risk.lsd.ornl.gov/rap hp.shtml.

• Weight of Evidence (WOE): IRIS Web site (http://www.epa.gov/IRIS).

• Oral no observable adverse effect level (NOAEL), inhalation NOAEL, oral lowest obserable adverse effect level (LOAEL) and inhalation LOAEL: IUCLID, 1996; HEAST, 1994; Kincaid and Geibig, 1998; EPA, 2000a; SRC, 2000; EPA, 2000b; Geibig and Swanson, 2000; Sax and Lewis, 1987; NIOSH, 1978; EPA, 1984; and EPA, 1987.

• Fish LC50 and fish NOAEL: EPA, 2001; HSDB; Davis et al. 1994, Appendix E; and Geiger et al., 1984, 1985, 1986, 1988, 1990.

CAS #	Material	Reason for Exclusion ^a
NA	ABS plastic	judgment
21645-51-2	Aluminium Hydroxide (Al(OH)3)	GRAS
10043-01-3	Aluminium Sulfate (Al2(SO4)3)	GRAS
10043-01-3	Aluminum Sulfate (Al2(SO4)3)	GRAS
7440-37-1	Argon (Ar gas)	judgment
1302-78-9	Bentonite (Al2O3.4SiO2.H2O, in ground)	judgment
NA	BOD (Biological Oxygen Demand)	judgment
106-97-8	Butane (n-C4H10)	GRAS
25167-67-3	Butene (1-CH3CH2CHCH2)	judgment
7440-70-2	Calcium (Ca)	judgment
No CAS #	Calcium (Ca++)	judgment
10043-52-4	Calcium Chloride (CaCl2)	judgment
1305-62-0	Calcium hydroxide [Ca(OH)2, hydrated lime]	judgment
7778-18-9	Calcium Sulfate	judgment
124-38-9	Carbon Dioxide (CO2)	judgment
NA	Carbonate ion [Carbonates (CO3, HCO3-, CO2]	judgment
NA	COD (Chemical Oxygen Demand)	judgment
16887-00-6	Chloride (Cl-)	judgment
1318-74-7	Clay (in ground)	judgment
NA	Dissolved solids	judgment
No CAS #	Dolomite (CaCO3.MgCO3, in ground)	judgment
26265-08-7	Epoxy resin (PC Board-epoxy resin)	judgment
141-78-6	Ethyl acetate (C4H8O2)	GRAS
64-17-5	Ethanol (Ethyl Alcohol)	GRAS
7705-08-0	Ferric chloride (FeCl3)	GRAS
NA	Ferrite	judgment
No CAS #	Glass	judgment
7440-59-7	Helium (He)	GRAS
NA	Nonmethane hydrocarbons	judgment
NA	Hydrocarbons (unspecified)	judgment
1333-74-0	Hydrogen gas (H2)	judgment
14380-61-1	Hypochlorite (ClO-)	judgment
20461-54-5	Iodide (I-)	judgment
7553-56-2	Iodine (I)	judgment
7439-89-6	Iron (Fe)	judgment
No CAS #	Iron (Fe++, Fe3+)	judgment
7720-78-7	Iron Sulfate (FeSO4, ore)	judgment
8008-20-6	Kerosene	judgment

 Table K-9. List of Materials Excluded from Toxic Classification

CAS #	Material	Reason for Exclusion ^a
7439-90-9	Krypton Gas	judgment
No CAS #	Lignite (in ground)	judgment
1305-78-8	Lime	judgment
471-34-1	Limestone (CaCO3, in ground)	judgment
7439-95-4	Magnesium (Mg)	judgment
No CAS #	Magnesium cmpds [Magnesium (Mg++)]	judgment
7440-01-9	Neon	judgment
7727-37-9	Nitrogen	GRAS
74-98-6	n-propane [Propane (C3H8)]	GRAS
NA	Oil & grease	judgment
No CAS #	Olivine ((Mg,Fe)2SiO4, ore)	judgment
144-62-7	Oxalic Acid (C2H2O4)	judgment
7782-44-7	Oxygen (O2)	judgment
NA	Phosphates (PO4-3)	judgment
9011-87-4	Poly(methyl methacrylate) [PMMA (Acrylic resin)]	judgment
25971-63-5	Polycarbonate resin	judgment
NA	Polycyclic Aromatic Hydrocarbons (PAH, unspecified)	judgment
9002-88-4	polyethylene (PE) foam, cushion	judgment
No CAS #	Polyimide Resin	judgment
9003-53-6	Polystyrene [Styrene, polymer (C8H8)]	judgment
7440-09-7	Potassium (K)	judgment
No CAS #	Potassium (K+)	judgment
584-08-7	Potassium carbonate (K2CO3)	judgment
7447-40-7	Potassium Chloride (KCl, as K2O, in ground)	judgment
79-09-4	Propionic Acid (CH3CH2COOH)	GRAS
1332-09-8	Pumice	judgment
1309-36-0	Pyrite (FeS2, ore)	judgment
14808-60-7	Silica sand [Silicon dioxide (SiO2)]	GRAS
7440-23-5	Sodium (Na)	judgment
No CAS #	Sodium (Na+)	judgment
497-19-8	Sodium carbonate (Na2CO3, soda ash)	judgment
7647-14-5	Sodium Chloride (NaCl, in ground or in sea)	GRAS
1310-73-2	Sodium hydroxide (NaOH)	judgment
9003-55-8	Styrene-butadiene copolymers (C12H14)	judgment
14808-79-8	Sulfates (SO4)	judgment
18496-25-8	Sulfides (S)	judgment
14265-45-3	Sulfites (SO3)	judgment
7704-34-9	Sulfur	judgment

Table K-9. List of Materials Excluded from Toxic Classification

Table K-7. List of Water has Excluded from Toxic Classification						
CAS #	Material	Reason for Exclusion ^a				
NA	Suspended Solids	judgment				
14807-96-6	Talcum (4SiO2.3MgO.H2O, ore)	judgment				
No CAS #	TOCs (Total organic compounds)	judgment				

Table K-9. List of Materials Excluded from Toxic Classification

^a NOTES:

(1) GRAS = Generally Regarded as Safe by the U.S. Food and Drug Administration.

(2) Some materials were excluded based on judgement if they are nutrients: calcium, chloride, iodine, iron, magnesium, phosphorous, potassium, sodium (per the Risk Assessment Guidance for Superfund [RAGS], EPA/540/1-89/002, December 1989 and the RAGS Region IV update).

(3) This list was reviewed by the U.S. EPA DFE Workgroup (Appendix C, Table C-2).

Chemical	CAS #	Oral Slope Factor (mg/kg-day) ⁻¹	Inhalation Slope Factor (mg/kg-day) ⁻¹
Acephate	30560-19-1	8.70E-03	
Acetaldehyde	75-07-0		7.70E-03
Acrylamide	79-06-1	4.50E+00	4.50E+00
Acrylonitrile	107-13-1	5.40E-01	2.40E-01
Alachlor	15972-60-8	8.00E-02	
Aldrin	309-00-2	1.70E+01	1.70E+01
Aniline	62-53-3	5.70E-03	
Aramite	140-57-8	2.50E-02	2.50E-02
Aroclor 1016	12674-11-2	4.00E-01	4.00E-01
Aroclor 1016	12674-11-2	2.00E+00	2.00E+00
Aroclor 1221	11104-28-2	4.00E-01	4.00E-01
Aroclor 1221	11104-28-2	2.00E+00	2.00E+00
Aroclor 1232	11141-16-5	4.00E-01	4.00E-01
Aroclor 1232	11141-16-5	2.00E+00	2.00E+00
Aroclor 1242	53469-21-9	4.00E-01	4.00E-01
Aroclor 1242	53469-21-9	2.00E+00	2.00E+00
Aroclor 1248	12672-29-6	4.00E-01	4.00E-01
Aroclor 1248	12672-29-6	2.00E+00	2.00E+00
Aroclor 1254	11097-69-1	4.00E-01	4.00E-01
Aroclor 1254	11097-69-1		2.00E+00
Aroclor 1260	11096-82-5	4.00E-01	4.00E-01
Aroclor 1260	11096-82-5	2.00E+00	2.00E+00
Arsenic, Inorganic	7440-38-2	1.50E+00	5.00E+01
Atrazine	1912-24-9	2.22E-01	
Azobenzene	103-33-3	1.10E-01	1.10E-01
Benz[a]anthracene	56-55-3	7.30E-01	3.10E-01
Benzene	71-43-2	5.50E-02	2.90E-02
Benzidine	92-87-5	2.30E+02	2.30E+02
Benzo[a]pyrene	50-32-8	7.30E+00	3.10E+00
Benzo[b]fluoranthene	205-99-2	7.30E-01	3.10E-01
Benzo[k]fluoranthene	207-08-9	7.30E-02	3.10E-02
Benzotrichloride	98-07-7	1.30E+01	
Benzyl Chloride	100-44-7	1.70E-01	
Beryllium and compounds	7440-41-7	4.30E+00	8.40E+00
Bis(2-chloro-1-methylethyl)ether (Technical)	108-60-1	7.00E-02	3.50E-02
Bis(2-chloroethyl)ether	111-44-4	1.10E+00	1.10E+00
Bis(2-ethylhexyl)phthalate	117-81-7	1.40E-02	

carcinogenic nazaru value								
Chemical	CAS #	Oral Slope Factor (mg/kg-day) ⁻¹	Inhalation Slope Factor (mg/kg-day) ⁻¹					
Bis(chloromethyl)ether	542-88-1	2.20E+02	2.20E+02					
Bromodichloromethane	75-27-4	6.20E-02						
Bromoform	75-25-2	7.90E-03	3.90E-03					
Butadiene, 1,3-	106-99-0		1.80E+00					
Cadmium (Diet)	7440-43-9		6.10E+00					
Cadmium (Water)	7440-43-9		6.10E+00					
Captafol	2425-06-1	8.60E-03						
Captan	133-06-2	3.50E-03						
Carbazole	86-74-8	2.00E-02						
Carbon Tetrachloride	56-23-5	1.30E-01	5.30E-02					
Chloranil	118-75-2	4.03E-01						
Chlordane	057-74-9	3.50E-01	1.30E+00					
Chloro-2-methylaniline HCl, 4-	3165-93-3	4.60E-01						
Chloro-2-methylaniline, 4-	95-69-2	5.80E-01						
Chlorobenzilate	510-15-6	2.70E-01	2.70E-01					
Chlorodibromoethane	73506-94-2	8.40E-02						
Chloroform	67-66-3	6.10E-03	8.10E-02					
Chloromethane	74-87-3	1.30E-02	6.30E-03					
Chloronitrobenzene, o-	88-73-3	2.50E-02						
Chloronitrobenzene, p-	121-73-3	1.80E-02						
Chlorothalonil	1897-45-6	1.10E-02						
Chromium VI (chromic acid mists)	18540-29-9		4.10E+01					
Chromium VI (particulates)	18540-29-9		4.10E+01					
Chrysene	218-01-9	7.30E-03	3.10E-03					
Coke Oven Emissions	8007-45-2		2.20E+00					
Crotonaldehyde, trans-	123-73-9	1.90E+00						
Cyanazine	21725-46-2	8.40E-01						
Cyclohexane, 1,2,3,4,5-pentabromo-6-chloro-	87-84-3	2.30E-02						
DDD	72-54-8	2.40E-01						
DDE	72-55-9	3.40E-01						
DDT	50-29-3	3.40E-01	3.40E-01					
Di(2-ethylhexyl)adipate	103-23-1	1.20E-03						
Diallate	2303-16-4	6.10E-02						
Dibenz[a,h]anthracene	53-70-3	7.30E+00	3.10E+00					
Dibromo-3-chloropropane, 1,2-	96-12-8	1.40E+00	2.40E-03					
Dibromochloromethane	124-48-1	8.40E-02						
Dibromoethane, 1,2-	106-93-4	8.50E+01	7.60E-01					

Chemical	CAS #	Oral Slope Factor (mg/kg-day) ⁻¹	Inhalation Slope Factor (mg/kg-day) ⁻¹
Dichloro-2-butene, 1,4-	764-41-0		9.30E+00
Dichlorobenzene, 1,4-	106-46-7	2.40E-02	
Dichlorobenzidine, 3,3'-	91-94-1	4.50E-01	
Dichloroethane, 1,2-	107-06-2	9.10E-02	9.10E-02
Dichloroethylene, 1,1-	75-35-4	6.00E-01	1.20E+00
Dichloropropane, 1,2-	78-87-5	6.80E-02	
Dichloropropene, 1,3-	542-75-6	1.00E-01	1.40E-02
Dichlorvos	62-73-7	2.90E-01	
Dieldrin	60-57-1	1.60E+01	1.60E+01
Diethylstilbesterol	56-53-1	4.70E+03	4.90E+02
Dimethoxybenzidine, 3,3'-	119-90-4	1.40E-02	
Dimethylaniline HCl, 2,4-	21436-96-4	5.80E-01	
Dimethylaniline, 2,4-	095-68-1	7.50E-01	
Dimethylbenzidine, 3,3'-	119-93-7	9.20E+00	
Dimethylhydrazine, 1,1-	57-14-7	3.00E+00	1.72E+01
Dinitrotoluene Mixture, 2,4/2,6-	25321-14-6	6.80E-01	
Dinitrotoluene, 2,4-	121-14-2	6.80E-01	
Dinitrotoluene, 2,6-	606-20-2	6.80E-01	
Dioxane, 1,4-	123-91-1	1.10E-02	
Diphenylhydrazine, 1,2-	122-66-7	8.00E-01	8.00E-01
Direct Black 38	1937-37-7	8.60E+00	
Direct Blue 6	2602-46-2	8.10E+00	
Direct Brown 95	16071-86-6	9.30E+00	
Epichlorohydrin	106-89-8	9.90E-03	4.20E-03
Ethyl Acrylate	140-88-5	4.80E-02	
Ethylbenzene	100-41-4		3.85E-03
Ethylene Oxide	75-21-8	1.02E+00	3.50E-01
Ethylene Thiourea	96-45-7	1.10E-01	
Folpet	133-07-3	3.50E-03	
Fomesafen	72178-02-0	1.90E-01	
Formaldehyde	50-00-0		4.50E-02
Furazolidone	67-45-8	3.80E+00	
Furium	531-82-8	5.00E+01	
Furmecyclox	60568-05-0	3.00E-02	
Heptachlor	76-44-8	4.50E+00	4.50E+00
Heptachlor Epoxide	1024-57-3	9.10E+00	9.10E+00
Hexachlorobenzene	118-74-1	1.60E+00	1.60E+00

Table K-10.	Chemicals used to calculate mean slope factor values for calculating
	carcinogenic hazard value

Chemical	CAS #	Oral Slope Factor (mg/kg-day) ⁻¹	Inhalation Slope Factor (mg/kg-day) ⁻¹
Hexachlorobutadiene	87-68-3	7.80E-02	7.80E-02
Hexachlorocyclohexane, Alpha-	319-84-6	6.30E+00	6.30E+00
Hexachlorocyclohexane, Beta-	319-85-7	1.80E+00	1.80E+00
Hexachlorocyclohexane, Gamma-	58-89-9	1.30E+00	
Hexachlorocyclohexane, Technical	608-73-1	1.80E+00	1.80E+00
Hexachlorodibenzo-p-dioxin, Mixture	19408-74-3	6.20E+03	4.55E+03
Hexachloroethane	67-72-1	1.40E-02	1.40E-02
Hexahydro-1,3,5-trinitro-1,3,5-triazine (RDX)	121-82-4	1.10E-01	
HpCDD, 2,3,7,8-	37871-00-4	1.50E+03	1.50E+03
HpCDF, 2,3,7,8-	38998-75-3	1.50E+03	1.50E+03
HxCDD, 2,3,7,8-	34465-46-8	1.50E+04	1.50E+04
HxCDF, 2,3,7,8-	55684-94-1	1.50E+04	1.50E+04
Hydrazine	302-01-2	3.00E+00	1.70E+01
Hydrazine Sulfate	10034-93-2	3.00E+00	1.70E+01
Indeno[1,2,3-cd]pyrene	193-39-5	7.30E-01	3.10E-01
Isophorone	78-59-1	9.50E-04	
Methoxy-5-nitroaniline, 2-	99-59-2	4.60E-02	
Methyl Hydrazine	60-34-4	3.00E+00	1.72E+01
Methyl-5-Nitroaniline, 2-	99-55-8	3.30E-02	
Methylaniline Hydrochloride, 2-	636-21-5	1.80E-01	
Methylene Chloride	75-09-2	7.50E-03	1.65E-03
Methylene-bis(2-chloroaniline), 4,4'-	101-14-4	1.30E-01	1.30E-01
Methylene-bis(N,N-dimethyl) Aniline, 4,4'-	101-61-1	4.60E-02	
Methylenebisbenzenamine, 4,4'-	101-77-9	2.50E-01	
Mirex	2385-85-5	1.80E+00	
Nickel Refinery Dust	NA		8.40E-01
Nickel Subsulfide	12035-72-2		1.70E+00
Nitrofurazone	59-87-0	1.50E+00	
Nitropropane, 2-	79-46-9	9.50E+00	9.40E+00
Nitrosodiethanolamine, N-	1116-54-7	2.80E+00	
Nitrosodiethylamine, N-	55-18-5	1.50E+02	1.50E+02
Nitrosodimethylamine, N-	62-75-9	5.10E+01	5.10E+01
Nitroso-di-N-butylamine, N-	924-16-3	5.40E+00	5.40E+00
Nitroso-di-N-propylamine, N-	621-64-7	7.00E+00	
Nitrosodiphenylamine, N-	86-30-6	4.90E-03	
Nitrosomethylethylamine, N-	10595-95-6	2.20E+01	
Nitroso-N-ethylurea, N-	759-73-9	1.40E+02	

Chemical	CAS#	Oral Slope Factor (mg/kg-day) ⁻¹	Inhalation Slope Factor (mg/kg-day) ⁻¹
Nitrosopyrrolidine, N-	930-55-2	2.10E+00	2.10E+00
OCDD	3268-87-9	1.50E+02	1.50E+02
OCDF	39001-02-0	1.50E+02	1.50E+02
PeCDD, 2,3,7,8-	36088-22-9	7.50E+04	7.50E+04
PeCDF, 1,2,3,7,8-	57117-41-6	7.50E+04	7.50E+04
PeCDF, 2,3,4,7,8-	57117-31-4	7.50E+03	7.50E+03
Pentachloronitrobenzene	82-68-8	2.60E-01	
Pentachlorophenol	87-86-5	1.20E-01	
Phenylenediamine, o-	95-54-5	4.70E-02	
Phenylphenol, 2-	90-43-7	1.94E-03	
Polybrominated Biphenyls	59536-65-1	8.90E+00	
Polychlorinated Biphenyls (high risk)	1336-36-3	2.00E+00	2.00E+00
Polychlorinated Biphenyls (low risk)	1336-36-3	4.00E-01	4.00E-01
Polychlorinated Biphenyls (lowest risk)	1336-36-3	7.00E-02	
Prochloraz	67747-09-5	1.50E-01	
Propylene Oxide	75-56-9	2.40E-01	1.30E-02
Quinoline	91-22-5	1.20E+01	
Simazine	122-34-9	1.20E-01	
Sodium Diethyldithiocarbamate	148-18-5	2.70E-01	
Stirofos (Tetrachlorovinphos)	961-11-5	2.40E-02	
TCDD, 2,3,7,8-	1746-01-6	1.50E+05	1.50E+05
TCDF, 2,3,7,8-	51207-31-9	1.50E+04	1.50E+04
Tetrachloroethane, 1,1,1,2-	630-20-6	2.60E-02	2.60E-02
Tetrachloroethane, 1,1,2,2-	79-34-5	2.00E-01	2.00E-01
Tetrachloroethylene	127-18-4	5.20E-02	2.00E-03
Tetrachlorotoluene, p- alpha, alpha, alpha-	5216-25-1	2.00E+01	
Toluene-2,4-diamine	95-80-7	3.20E+00	
Toluidine, o- (Methylaniline, 2-)	95-53-4	2.40E-01	
Toluidine, p-	106-49-0	1.90E-01	
Toxaphene	8001-35-2	1.10E+00	1.10E+00
Trichloroaniline HCl, 2,4,6-	33663-50-2	2.90E-02	
Trichloroaniline, 2,4,6-	634-93-5	3.40E-02	
Trichloroethane, 1,1,2-	79-00-5	5.70E-02	5.70E-02
Trichloroethylene	79-01-6	1.10E-02	6.00E-03
Trichlorophenol, 2,4,6-	88-06-2	1.10E-02	1.00E-02
Trichloropropane, 1,2,3-	96-18-4	7.00E+00	
Trifluralin	1582-09-8	7.70E-03	

Chemical	CAS #	Oral Slope Factor (mg/kg-day) ⁻¹	Inhalation Slope Factor (mg/kg-day) ⁻¹
Trimethyl Phosphate	512-56-1	3.70E-02	
Trinitrotoluene, 2,4,6-	118-96-7	3.00E-02	
Vinyl Bromide	593-60-2		1.10E-01
Vinyl Chloride	75-01-4	1.40E+00	3.08E-02
geometric mean		0.71	1.70
count (n)		175	105

blank=no data

Source: Risk Assessment Information System (RAIS), http://risk.lsd.ornl.gov/cgi-bin/tox/TOX_9801 (downloaded 11/00): IRIS/HEAST Slope Factors.

REFERENCES

- Davis, G. A., L. Kincaid, M. Swanson, T. Schultz, J. Bartmess, B. Griffith, and S. Jones. 1994. Chemical Hazard Evaluation for Management Strategies: A Method for Ranking and Scoring Chemicals by Potential Human Health and Environmental Impacts, Appendix E, United States Environmental Protection Agency, EPA/600/R-94/177, September.
- EPA (U. S. Environmental Protection Agency). 2001. EPA data search of the U.S. EPA Ecotox Search engine (http://www.epa.gov/ecotox) provided to University of Tennessee, January.
- EPA. 2000a. *Flexographic Ink Options: A Cleaner Technologies Substitutes Assessment (DRAFT)*. EPA/744-R-00-004A&B. Washington, D.C. November.
- EPA. 2000b. EPA data search provided to the University of Tennessee, December.
- EPA. 1994. *Health Effects Assessment Summary Tables. Annual Update*. Environmental Criteria and Assessment Office, Office of Health and Environmental Assessment, Office of Research and Development, Cincinnati, OH.
- EPA. 1987. *Health Effects Assessment for Dichlorobenzene*. Environmental Criteria and Assessment Office, Office of Health and Environmental Assessment, Office of Research and Development, Cincinnati, OH.
- EPA. 1984. *Health Affects Document for Carbon Tetrachloride*. EPA/600/8-82-001F. Environmental Criteria and Assessment Office, Office of Health and Environmental Assessment, Office of Research and Development, Cincinnati, OH.
- Geibig, J. and M. Swanson. 2000. Printed Wiring Board Surface Finishes Draft Cleaner Technologies Substitutes Assessment. University of Tennessee. EPA 744-R-00-013B, December.
- Geiger, D. L., L. T. Brooke, D. J. Call and C. E. Northcott, eds. 1984. Acute Toxicities of Organic Chemicals to Fathead Minnows (Pimephales promelas), Vol. 1. Center for Lake Superior Environmental Studies, University of Wisconsin, Superior, WI.
- Geiger, D. L., L. T. Brooke, D. J. Call and C. E. Northcott, eds. 1985. *Acute Toxicities of Organic Chemicals to Fathead Minnows (Pimephales promelas)*, Vol. 2. Center for Lake Superior Environmental Studies, University of Wisconsin, Superior, WI.
- Geiger, D. L., L. T. Brooke, D. J. Call and C. E. Northcott, eds. 1986. Acute Toxicities of Organic Chemicals to Fathead Minnows (Pimephales promelas), Vol. 3. Center for Lake Superior Environmental Studies, University of Wisconsin, Superior, WI.

- Geiger, D. L., L. T. Brooke, D. J. Call and C. E. Northcott, eds. 1988. *Acute Toxicities of Organic Chemicals to Fathead Minnows (Pimephales promelas)*, Vol. 4. Center for Lake Superior Environmental Studies, University of Wisconsin, Superior, WI.
- Geiger, D. L., L. T. Brooke, D. J. Call and C. E. Northcott, eds. 1990. *Acute Toxicities of Organic Chemicals to Fathead Minnows (Pimephales promelas)*, Vol. 5. Center for Lake Superior Environmental Studies, University of Wisconsin, Superior, WI.
- HSDB (Hazardous Substances Data Bank). National Library of Medicine, Toxicology Data Network (TOXNET) (http://toxnet.nlm.nih.gov/cgi-bin/sis/htmlgen?HSDB).
- IUCLID (International Uniform Chemical Information Database). 1996. European Chemical Bureau, European Commission JRC, Environment Institute, Ispra, Italy.
- Kincaid, L. E. and J. R. Geibig. 1998. Printed Wiring Board Cleaner Technologies Substitutes Assessment: Making Holes Conductive. University of Tennessee. EPA 744-R-98-004a,b, August.
- NIOSH (National Institute of Occupational Safety and Health). 1978. Occupational Health Guideline for Carbon Monoxide.
- Sax, N. I and R. J. Lewis. 1987. *Hazardous Chemicals Desk Reference*. VanNostrand Reinhold, New York.
- SRC (Syracuse Research Corporation). 2000. SRC data search provided to the University of Tennessee, December.

APPENDIX L

APPENDIX L

SECONDARY ISSUES TO THE USE LIFE-CYCLE STAGE OF COMPUTER DISPLAYS

this page intentionally left blank

APPENDIX L

SECONDARY ISSUES TO THE USE LIFE-CYCLE STAGE OF COMPUTER DISPLAYS

1. ELECTRIC AND MAGNETIC FIELDS (EMFs)

Electric and magnetic fields (EMFs) are invisible lines of force that surround any electrical device, including power lines, electrical wiring, and electrical equipment. Electric fields are produced by voltage and increase in strength as the voltage increases. Magnetic fields result from the flow of current through wires and or electrical devices and increase in strength as the current increases. Most electrical equipment has to be turned on for a magnetic field to be produced, but electric fields are present even when equipment is switched off as long as it is connected to an electric power source. Electric fields are weakened or shielded by materials that conduct electricity (including human skin). Magnetic fields, on the other hand, pass through most materials and are therefore more difficult to shield and of greater concern. Both electric and magnetic fields decrease with distance from the source (NIOSH, et al 1996).

Most information on EMFs from video display units (VDUs) pertains to CRT monitors. The following is excerpted from the World Health Organization fact sheet, "Video Display Units (VDUs) and Human Health (1998):"

"The typical VDU creates images in a large evacuated cathode-ray-tube (CRT) by directing a beam of high-energy electrons from the cathode onto a special phosphorcoated glass screen. This coating emits light when struck by the fast-moving electrons. The electron beam creates the image from computer signals that control coils, at the back of the CRT, that sweep the electrons in the vertical and horizontal directions. These coils are called vertical and horizontal deflection coils. The electronic circuitry used to create the image gives rise to static electric and magnetic fields, as well as low and high frequency electromagnetic fields...

Electric and magnetic fields are emitted in three different frequency ranges. The horizontal deflection coils emit fields operating predominantly in the frequency range 15-35kHz. Extremely low frequency (ELF) fields at 50 to 60 Hz come from the power supply, transformers and the vertical deflection coils. Finally, weak signals at higher radio frequencies (RF) come from the VDU's interior electronic circuitry and signals received from the computer."

Very little information was found on the relative magnitude of EMFs emitted by CRTs and LCDs. However, typical household power operates at 50-60 Hz and 120 volts – these remain relatively constant no matter how much power in watts a piece of electrical equipment needs or draws. What fluctuates with power demand is current (in amperes, or amps). CRTs consume a greater quantity of power in watts, and therefore require a larger amount of electrical current in amps. Since magnetic fields increase in strength with increased current, it is assumed a CRT will

APPENDIX L

generate a larger magnetic field than an LCD. Additionally, some of the components discussed above, such as the horizontal and vertical deflection coils, are found in CRTs but not in LCDs. Thus, due to their power handling needs and capabilities, CRTs also generate EMFs that are not generated by LCDs. However, according to NoRad Corporation, a manufacturer and marketer of EMF shielding products for monitors, it is a common misconception that LCDs do not emit EMFs because of their smaller current draw; backlit LCD displays can emit significant levels of both magnetic fields and electric fields (NoRad, undated).

To address concerns about potential health effects from EMF exposure, in 1992 the U.S. Congress authorized the Electric and Magnetic Fields Research and Public Information Program (EMF-RAPID Program) and directed the National Institute of Environmental Health Sciences (NIEHS) and the Department of Energy (DOE) to direct and manage a program of research and analysis aimed at providing scientific evidence to clarify the potential for health risks from exposure to ELF-EMF (NIEHS, 1999). After several years of research, in 1999 NIEHS issued a report to Congress on the health effects from exposure to power-line frequency EMFs, which concluded the following (NIEHS, 1999):

"The scientific evidence suggesting that ELF-EMF exposures pose any health risk is weak. The strongest evidence for health effects comes from associations observed in human populations with two forms of cancer: childhood leukemia and chronic lymphocytic leukemia in occupationally exposed adults. While the support from individual studies is weak, the epidemiological studies demonstrate, for some methods of measuring exposure, a fairly consistent pattern of a small, increased risk with increasing exposure that is somewhat weaker for chronic lymphocytic leukemia than for childhood leukemia. In contrast, the mechanistic studies and the animal toxicology literature fail to demonstrate any consistent pattern across studies although sporadic finding of biological effects (including increased cancers in animals) have been reported. No indication of increased leukemia in experimental animals has been observed. The lack of connection between the human data and the experimental data (animal and mechanistic) severely complicates the interpretation of these results. The human data are in the "right" species, are tied to "real-life" exposures and show some consistency that is difficult to ignore. This assessment is tempered by the observation that given the weak magnitude of these increased risks, some other factor or common source of error could explain these findings. However, no consistent explanation other than exposure to ELF-EMF has been identified.

Epidemiological studies have serious limitations in their ability to demonstrate a cause and effect relationship whereas laboratory studies, by design, can clearly show that cause and effect are possible. Virtually all of the laboratory evidence in animals and humans and most of the mechanistic work done in cells fail to support a causal relationship between exposure to ELF-EMF at environmental levels and changes in biological function or disease status. The lack of consistent, positive findings in animal or mechanistic studies weakens the belief that this association is actually due to ELF-EMF, but it cannot completely discount the epidemiological findings.

The NIEHS concludes that ELF-EMF exposure cannot be recognized as entirely safe because of weak scientific evidence that exposure may pose a leukemia hazard. In our

opinion, this finding is insufficient to warrant aggressive regulatory concern. However, because virtually everyone in the United States uses electricity and therefore is routinely exposed to ELF-EMF, passive regulatory action is warranted such as a continued emphasis on educating both the public and the regulated community on means aimed at reducing exposures. The NIEHS does not believe that other cancers on non-cancer health outcomes provide sufficient evidence of a risk to currently warrant concern."

More recently, an expert scientific working group of the Monographs Programme of the International Agency for Research on Cancer (IARC) released its findings from a review of health effects of static and ELF EMFs. IARC concluded that ELF magnetic fields are possibly carcinogenic to humans, based on consistent statistical associations of high level residential magnetic fields with a doubling of risk of childhood leukemia. However, IARC also concluded that children who are exposed to residential ELF magnetic fields less than 0.4 microTesla (4 milligauss)¹ have no increased risk of leukemia (IARC, 2001). To help put this in perspective, Table L-1 presents the average magnetic field exposures for clerical workers with and without computers.² As shown in the table, clerical workers with computers have increased average daily exposures of about 0.07 μ T (0.7 mG) over clerical workers without computers. The average daily median for workers with computers is 0.12 μ T (1.2 mG) and the exposure range is 0.05 to 0.45 μ T (0.5 to 4.5 mG). Only the upper end of the exposure range exceeds the exposure level that IARC concluded has an increased risk of childhood leukemia (0.4 μ T).

Table L-1. Average magnetic field exposures for clerical workers*

Type of worker	Average daily median (µT)	Exposure range (μT)
Clerical workers with computers	0.12	0.05 to 0.45
Clerical workers without computers	0.05	0.02 to 0.2

Source: NIOSH Fact Sheet: EMFs in the Workplace

* The source does not give the distance at which measurements were taken. Monitor EMF emission measurements are often taken at a distance of 30 cm (approximately 12 inches) or 50 cm (approximately 20 inches).

In summary, no data were found on EMF measurements from LCDs. However, because of the lesser current requirements of LCDs compared to CRTs, it is assumed that LCDs also generate a lesser magnetic field. NIEHS has concluded that the evidence that ELF-EMF exposures pose any health risk is weak. However, since publication of the NIEHS report, IARC has classified ELF-EMFs as possibly carcinogenic to humans. Based on the data in Table L-3, it

¹Magnetic field intensity is measured in units of tesla (T) or gauss (G). One tesla equals 10,000 gauss. Since most environmental EMF exposure involve magnetic field intensities that are only a fraction of a tesla or a gauss, they are commonly measured in units of microteslas (μ T) or milligauss (mG). One μ T is equal to 10 mG.

² Given the fact that desktop LCD monitors were only recently introduced into the marketplace, it is assumed that the computers used by clerical workers for whom measurements were taken were equipped with CRT monitors, although the data could also include clerical workers who used laptops with LCD displays.

APPENDIX L

appears most exposures to ELFs-EMFs from computer displays may be below the carcinogenicity concern level determined by IARC (0.4 μ T).

2. ERGONOMIC ISSUES

Merriam-Webster's Online Dictionary defines ergonomics as "an applied science concerned with designing and arranging things people use so that the people and things interact most efficiently and safely – also called human engineering." While CRT and LCD desktop monitors are both usable in the same environments (in most cases), there are differences in their sizes and the way they present information to the user in their working environment. Thus the potential exists for there to be differences in the way a user might physically interact with their CRT or LCD monitor. (Note that Eye Strain is addressed as a separate issue in this Appendix [see L.3], and issues related to image generation on the monitor surface are addressed there.)

In reviewing several documents on ergonomics and the placement of computer monitors in or around a user's working area, some sources discuss the differences in using a CRT versus an LCD monitor. It is assumed in most discussions that the user is able to move/adjust the monitor to within a recommended operating position for use (e.g., eye-to-screen distance, vertical monitor location [with respect to horizontal eye level]). Thus, with unlimited resources within which to setup a computing environment, it is expected that only viewing angle would potentially favor a CRT over an LCD. CRTs provide a horizontal viewing angle of 180° whereas early-model LCD monitors were limited to an almost straight on view. The most recent LCDs on the market have a 120° capability (EIZO 1999). Although less than the CRT horizontal viewing angle, these current viewing angles significantly diminish the differences in viewing ability between the two monitor types. The reduced viewing angle of the LCD appears to mainly be a factor when presenting information to users who are viewing the screen from the side, such as during presentations to multiple users.

However, the user's environment is often limited by physical constraints on where the monitor can be placed and user seated. With this in mind, the physical size and footprint of the monitor becomes more of an ergonomics issue. In cases where the footprint of a monitor dictates that the monitor not go directly in-front of the user, but to either the left or right side, an increased level of ergonomic stress may be realized (typically neck twist) if the user is unable to adjust his or her seated position accordingly. Even the smaller footprint CRTs being manufactured today have a much larger footprint than those of closely-equivalent LCD monitors, typically occupying almost six times the depth of LCDs (IBM 1999). Additionally, some LCD monitors can be wall-mounted, almost completely removing the monitor from the desktop. Thus, it appears that the smaller footprint of the LCD may offer benefits in more easily positioning the monitor for optimum user use.

3. EYE STRAIN

There are numerous sensations that can be interpreted to be eye strain in the use of monitors, including but not limited to burning, tightness, sharp pains, dull pains, watering, blurring, double vision and headaches. Many of the principal factors that cause eye strain can be corrected or improved by adequately setting up the computing environment (e.g., controlling the distance between the eyes and the screen, wearing corrective lenses, if needed, etc.). Others can be improved by adjusting the controls on the monitor (e.g., setting brightness or contrast). Of the factors identified in this study, glare and screen flicker are the two that appear to be most affected by the technology choice (e.g., CRT or LCD).

The distance between your eyes and the monitor should be at least 25" (Ankrum 1996), and using a CRT versus an LCD should have no effect on the necessary distance needed to clearly see the monitor screen, as long as the user is using an appropriately sized resolution for his or her monitor size. Both CRTs and LCDs should be equally readable, not considering the viewing of an LCD from outside its particular viewing angle. If the brightness levels of a CRT and LCD are set appropriately, then the contrast between what is being looked at and its immediate environment should be no different for the two monitor types. Lastly, a user has to have good or corrected vision before either monitor will be useful.

The only notable difference between CRTs and LCDs with respect to glare is their flatness. As LCDs have completely flat monitor surfaces or screens, versus most CRTs which have rounded screens, they significantly reduce the probability of reflected glare from overhead or nearby lights. There are two main types of CRT tube technology, shadow mask and aperture grill. Shadow masks are utilized in older, less flat monitors, and aperture grills are utilized in the more current variations that have flatter outside faces (IBM 1999). Flatter screens provide less opportunity for the occurrence of reflected glare.

CRTs, especially older models, may be prone to screen flicker, which contributes significantly to eye strain. CRT's use phosphor that has been excited by an electron beam to create light. After the phosphor is excited, it begins to decay. The electron beam needs to return to the phosphor in a specific amount of time to keep the phosphor from decaying to the point that the human eye can perceive it. The rate at which the electron beam returns to any given phosphor is called the refresh rate. If the refresh rate is too low, the decaying of the phosphor may be perceptible to the human eye as a flickering screen image. Because LCDs do not use phosphors to create the image they do not have a refresh rate and do not flicker.

REFERENCES

- Ankrum, D. R. 1996. "Monitor Location: Guidelines for Monitor Placement and Lighting." Web site available at http://ur-net.com/office-ergo/setting.htm.
- IARC (International Agency for Research on Cancer). 2001. "IARC Finds Limited Evidence that Residential Magnetic Fields Increase Risk of Childhood Leukemia." Press release accessed at http://www.iarc/fr/pageroot/preleases/pr136a.html on August 30, 2001. June 27.
- NIEHS (National Institute of Environmental Health Sciences). 1999. *NIEHS Report on Health Effects from Exposure to Power-Line Frequency Electric and Magnetic Fields*. Available as a pdf document from the following Web site http://www.niehs.nih.gov/emfrapid/home.htm. May.
- NIOSH (National Institute for Occupational Safety and Health), NIEHS, and DOE (Department of Energy). 1996. "Questions and Answers, EMF in the Workplace: Electric and Magnetic Fields Associated with the Use of Electric Power." Web site accessed at http://www.niehs.nih.gov/emfrapid/html/Q&A-Workplace.html. August 31, 2001. September.
- NIOSH. Undated. "NIOSH Fact Sheet: EMFs in the Workplace." Web site accessed at http://www.cdc.gov/niosh/emf2.html. August 30, 2001.
- NoRad. Undated. "Video Display Terminals as a Source of Electromagnetic Fields." Web site accessed at http://www.noradcorp.com. August 31, 2001.
- Poynton, C. 1997. "Reducing eyestrain from video and computer monitors." Web site available at http://www.inforamp.net/~poynton/notes/reducing_eyestrain/index.html.

APPENDIX M

LIFE-CYCLE IMPACT ASSESSMENT RESULTS

this page intentionally left blank

APPENDIX M

LIFE-CYCLE IMPACT ASSESSMENT RESULTS

Table M-1. CRT LCIA Results for the Renewable Resource Use Impact Category

Process Group	Material	LCI Data Type	Renewable Resource Use (kg)	% of Total
Materials Processing Life-cycle St	tage			
Steel Prod., cold-rolled, semi-finished	Water	secondary	4.69e+02	3.57e+00
Invar	Water	secondary	3.96e+01	3.02e-01
Lead	Water	secondary	1.40e+01	1.07e-01
Polycarbonate Production	Water	secondary	1.30e+01	9.93e-02
Ferrite mfg.	Water	secondary	1.11e+01	8.46e-02
ABS Production	Water	secondary	3.94e+00	3.00e-02
Styrene-butadiene Copolymer Prod.	Water	secondary	2.58e+00	1.97e-02
Polystyrene Prod., high-impact	Water	secondary	8.16e-01	6.22e-03
Polycarbonate Production	Sodium chloride (NaCl, in ground or in sea)	secondary	7.30e-01	5.56e-03
Steel Prod., cold-rolled, semi-finished	Limestone (CaCO3, in ground)	secondary	6.72e-01	5.12e-03
Aluminum Prod.	Limestone (CaCO3, in ground)	secondary	9.25e-02	7.05e-04
Lead	Limestone (CaCO3, in ground)	secondary	6.63e-02	5.05e-04
Lead	Sand (in ground)	secondary	4.79e-02	3.65e-04
Aluminum Prod.	Sodium chloride (NaCl, in ground or in sea)	secondary	2.56e-02	1.95e-04
Invar	Limestone (CaCO3, in ground)	secondary	9.08e-03	6.91e-05
ABS Production	Limestone (CaCO3, in ground)	secondary	7.62e-03	5.80e-05
Polycarbonate Production	Limestone (CaCO3, in ground)	secondary	6.00e-03	4.57e-05
Aluminum Prod.	Sand (in ground)	secondary	5.26e-03	4.00e-05
Polystyrene Prod., high-impact	Limestone (CaCO3, in ground)	secondary	3.78e-03	2.88e-05
Invar	Sand (in ground)	secondary	3.67e-03	2.80e-05
ABS Production	Gravel/Sand	secondary	3.60e-03	2.74e-05
Aluminum Prod.	Clay (in ground)	secondary	2.79e-03	2.13e-05
ABS Production	Sodium chloride (NaCl, in ground or in sea)	secondary	2.63e-03	2.00e-05
Steel Prod., cold-rolled, semi-finished	Gravel/Sand	secondary	2.05e-03	1.56e-05
Ferrite mfg.	Limestone (CaCO3, in ground)	secondary	1.58e-03	1.21e-05
Styrene-butadiene Copolymer Prod.	Sodium chloride (NaCl, in ground or in sea)	secondary	1.16e-03	8.82e-06
Styrene-butadiene Copolymer Prod.	Limestone (CaCO3, in ground)	secondary	1.13e-03	8.63e-06
Steel Prod., cold-rolled, semi-finished	Sodium chloride (NaCl, in ground or in sea)	secondary	1.02e-03	7.74e-06
Ferrite mfg.	Sand (in ground)	secondary	9.35e-04	7.12e-06
Invar	Clay (in ground)	secondary	8.67e-04	6.60e-06
Invar	Sodium chloride (NaCl, in ground or in sea)	secondary	6.95e-04	5.29e-06
Invar	Gravel/Sand	secondary	5.91e-04	4.50e-06
Ferrite mfg.	Gravel/Sand	secondary	5.77e-04	4.39e-06
Steel Prod., cold-rolled, semi-finished	Clay (in ground)	secondary	4.45e-04	3.39e-06
Polycarbonate Production	Sand (in ground)	secondary	3.79e-04	2.88e-06
Polystyrene Prod., high-impact	Sodium chloride (NaCl, in ground or in sea)	secondary	3.17e-04	2.42e-06
Ferrite mfg.	Sodium chloride (NaCl, in ground or in sea)	secondary	2.65e-04	2.02e-06
ABS Production	Sand (in ground)	secondary	2.54e-04	1.93e-06
Ferrite mfg.	Clay (in ground)	secondary	1.91e-04	1.45e-06
ABS Production	Clay (in ground)	secondary	1.25e-04	9.55e-07
Styrene-butadiene Copolymer Prod.	Sand (in ground)	secondary	6.21e-05	4.73e-07
Invar	Maize	secondary	5.75e-05	4.38e-07

Process Group	Material	LCI Data Type	Renewable	% of Total
Ferrite mfa	Maize	secondary	Resource Use (kg)	4 28e-07
Staal Prod. cold rollad somi finished	Sand (in ground)	secondary	5.340.05	4.070.07
Palvarbanata Production	Clay (in ground)	secondary	1.620.05	4.076-07
Stymona bytadiana Canalyman Drad	Clay (in ground)	secondary	4.020-05	1.860.07
Styrene-butadiene Copolymer Prod.	Ciay (in ground)	secondary	2.446-03	1.806-07
Polystyrene Prod., nign-impact	Sand (in ground)	secondary	2.420-05	1.846-07
Invar	Potatoes	secondary	1.556-05	1.18e-07
Ferrite mig.		secondary	1.51e-05	1.15e-07
Polycarbonate Production	Gravel/Sand	secondary	2.7/e-06	2.11e-08
Styrene-butadiene Copolymer Prod.	Gravel/Sand	secondary	8.2/e-0/	6.30e-09
Total Materials Processing			5.56e+02	4.23e+00
Manufacturing Life-cycle Stage				
LPG Production	Water, unspecified	secondary	1.04e+04	7.94e+01
CRT tube mfg.	Water	primary	8.11e+02	6.18e+00
Japanese Electric Grid	Water	model/secondary	4.43e+01	3.37e-01
PWB Mfg.	Water	model/secondary	4.22e+01	3.21e-01
Fuel Oil #6 Prod.	Water, unspecified	secondary	3.56e+01	2.71e-01
Glass/frit	Water	primary	3.51e+01	2.67e-01
Fuel Oil #2 Prod.	Water, unspecified	secondary	2.93e+01	2.23e-01
US electric grid	Water	model/secondary	1.82e+01	1.38e-01
Glass/frit	Sand	primary	2.40e+00	1.83e-02
Fuel Oil #4 Prod.	Water, unspecified	secondary	2.39e+00	1.82e-02
LPG Production	Limestone (CaCO3, in ground)	secondary	1.06e+00	8.10e-03
Natural Gas Prod.	Water	secondary	2.12e-01	1.61e-03
US electric grid	Limestone	model/secondary	3.85e-02	2.93e-04
Japanese Electric Grid	Limestone	model/secondary	3.06e-02	2.33e-04
LPG Production	Sand (in ground)	secondary	2.72e-02	2.08e-04
LPG Production	Sodium chloride (NaCl, in ground or in sea)	secondary	1.25e-02	9.52e-05
Fuel Oil #6 Prod.	Limestone (CaCO3, in ground)	secondary	8.56e-03	6.52e-05
Fuel Oil #2 Prod.	Limestone (CaCO3, in ground)	secondary	3.34e-03	2.54e-05
Natural Gas Prod.	Limestone (CaCO3, in ground)	secondary	1.84e-03	1.40e-05
Fuel Oil #4 Prod.	Limestone (CaCO3, in ground)	secondary	3.55e-04	2.71e-06
Fuel Oil #6 Prod.	Sand (in ground)	secondary	9.22e-05	7.02e-07
Fuel Oil #2 Prod.	Sand (in ground)	secondary	7.65e-05	5.82e-07
Fuel Oil #6 Prod	Sodium chloride (NaCl_in ground or in sea)	secondary	4 23e-05	3 22e-07
Fuel Oil #2 Prod	Sodium chloride (NaCl, in ground or in sea)	secondary	3 51e-05	2.67e-07
Fuel Oil #4 Prod	Sand (in ground)	secondary	6.21e-06	4 73e-08
Fuel Oil #4 Prod	Sodium chloride (NaCl_in ground or in sea)	secondary	2.85e-06	2.17e-08
Natural Gas Prod	Sand (in ground)	secondary	5.51e-07	4 19e-09
Natural Gas Prod	Sodium chloride (NaCl, in ground or in sea)	secondary	2.53e-07	1.92e-09
Total Manufacturing	Sourdin emoride (Waer, in ground of in sea)	secondary	1 150+04	8 720+01
Use Maintenance and Renair Life	-ovelo Stago		1.130+04	0.720101
US alastria grid	Water	modol/socondam/	1 140+02	8 67 2±00
US electric grid	Water Limestone	model/secondary	1.14e+03	1.820.02
		model/secondary	2.416+00	1.856-02
Total Use, Maintenance and Repa	lir.		1.14e+03	8.69e+00
End-of-life Life-cycle Stage				
CRT Incineration	Clay (in ground)	secondary	4.24e+00	3.23e-02
CRT landfilling	Clay (in ground)	primary	3.95e+00	3.01e-02
CRT Incineration	Sand (in ground)	secondary	1.40e+00	1.07e-02
CRT landfilling	Sand (in ground)	primary	1.31e+00	9.97e-03
US electric grid	Water	model/secondary	1.14e-01	8.67e-04

Table M-1. CRT LCIA Results for the Renewable Resource Use Impact Category

Process Group	Material	LCI Data Type	Renewable	% of Total
LPG Production	Water unspecified	secondary	Resource Use (Kg) 9.00e-02	6 86e-04
CRT landfilling	Limestone (CaCO3 in ground)	primary	6 32e-04	4 81e-06
US electric grid	Limestone	model/secondary	2.41e-04	1.84e-06
LPG Production	Limestone (CaCO3, in ground)	secondary	9.18e-06	6.99e-08
LPG Production	Sand (in ground)	secondary	2.35e-07	1.79e-09
LPG Production	Sodium chloride (NaCl, in ground or in sea)	secondary	1.08e-07	8.22e-10
Natural Gas Prod.	Sodium chloride (NaCl, in ground or in sea)	secondary	-1.27e-07	-9.65e-10
Natural Gas Prod.	Sand (in ground)	secondary	-2.76e-07	-2.10e-09
Fuel Oil #4 Prod.	Sodium chloride (NaCl, in ground or in sea)	secondary	-3.06e-05	-2.33e-07
Fuel Oil #4 Prod.	Sand (in ground)	secondary	-6.68e-05	-5.09e-07
Natural Gas Prod.	Limestone (CaCO3, in ground)	secondary	-9.22e-04	-7.02e-06
Fuel Oil #4 Prod.	Limestone (CaCO3, in ground)	secondary	-3.82e-03	-2.91e-05
Natural Gas Prod.	Water	secondary	-1.06e-01	-8.09e-04
CRT Incineration	Limestone (CaCO3, in ground)	secondary	-2.35e-01	-1.79e-03
CRT Incineration	Water	secondary	-1.73e+00	-1.32e-02
Fuel Oil #4 Prod.	Water, unspecified	secondary	-2.57e+01	-1.95e-01
Total End-of-Life			-1.66e+01	-1.27e-01
Total All Life-cycle Stages			1.31e+04	1.00e+02

 Table M-1. CRT LCIA Results for the Renewable Resource Use Impact Category

Process Group	Material	LCI Data Type	Renewable Resource Use (kg)	% of Total
Materials Processing Life-cycle S	Stage			
Steel Prod., cold-rolled, semi-finished	Water	secondary	2.30e+02	8.21e+00
Natural Gas Prod.	Water	secondary	1.48e+01	5.29e-01
PMMA Sheet Prod.	Water	secondary	8.86e+00	3.17e-01
Polycarbonate Production	Water	secondary	7.28e+00	2.60e-01
Styrene-butadiene Copolymer Prod.	Water	secondary	1.13e+00	4.03e-02
PET Resin Production	Water	secondary	8.19e-01	2.92e-02
Polycarbonate Production	Sodium chloride (NaCl, in ground or in sea)	secondary	4.07e-01	1.46e-02
Steel Prod., cold-rolled, semi-finished	Limestone (CaCO3, in ground)	secondary	3.29e-01	1.18e-02
Natural Gas Prod.	Limestone (CaCO3, in ground)	secondary	1.28e-01	4.59e-03
Aluminum Prod.	Limestone (CaCO3, in ground)	secondary	3.44e-02	1.23e-03
PMMA Sheet Prod.	Sodium chloride (NaCl, in ground or in sea)	secondary	1.92e-02	6.85e-04
Aluminum Prod.	Sodium chloride (NaCl, in ground or in sea)	secondary	9.54e-03	3.41e-04
PMMA Sheet Prod.	Limestone (CaCO3, in ground)	secondary	8.44e-03	3.01e-04
PMMA Sheet Prod.	Sand (in ground)	secondary	7.29e-03	2.60e-04
Polycarbonate Production	Limestone (CaCO3, in ground)	secondary	3.35e-03	1.20e-04
Aluminum Prod.	Sand (in ground)	secondary	1.96e-03	6.99e-05
PET Resin Production	Limestone (CaCO3, in ground)	secondary	1.91e-03	6.81e-05
Aluminum Prod.	Clay (in ground)	secondary	1.04e-03	3.71e-05
Steel Prod., cold-rolled, semi-finished	Gravel/Sand	secondary	1.01e-03	3.60e-05
Styrene-butadiene Copolymer Prod.	Sodium chloride (NaCl, in ground or in sea)	secondary	5.06e-04	1.81e-05
Steel Prod., cold-rolled, semi-finished	Sodium chloride (NaCl, in ground or in sea)	secondary	4.98e-04	1.78e-05
Styrene-butadiene Copolymer Prod.	Limestone (CaCO3, in ground)	secondary	4.96e-04	1.77e-05
Steel Prod., cold-rolled, semi-finished	Clay (in ground)	secondary	2.18e-04	7.78e-06
Polycarbonate Production	Sand (in ground)	secondary	2.11e-04	7.55e-06
PET Resin Production	Sodium chloride (NaCl, in ground or in sea)	secondary	1.27e-04	4.54e-06
Natural Gas Prod.	Sand (in ground)	secondary	3.85e-05	1.37e-06
Styrene-butadiene Copolymer Prod.	Sand (in ground)	secondary	2.71e-05	9.69e-07
Steel Prod., cold-rolled, semi-finished	Sand (in ground)	secondary	2.62e-05	9.35e-07
Polycarbonate Production	Clay (in ground)	secondary	2.58e-05	9.21e-07
Natural Gas Prod.	Sodium chloride (NaCl, in ground or in sea)	secondary	1.77e-05	6.30e-07
Styrene-butadiene Copolymer Prod.	Clay (in ground)	secondary	1.07e-05	3.81e-07
PET Resin Production	Sand (in ground)	secondary	5.81e-06	2.07e-07
PMMA Sheet Prod.	Clay (in ground)	secondary	4.60e-06	1.64e-07
Polycarbonate Production	Gravel/Sand	secondary	1.55e-06	5.53e-08
PMMA Sheet Prod.	Gravel/Sand	secondary	1.15e-06	4.11e-08
Styrene-butadiene Copolymer Prod.	Gravel/Sand	secondary	3.62e-07	1.29e-08
PET Resin Production	Clay (in ground)	secondary	9.08e-08	3.24e-09
PET Resin Production	Gravel	secondary	9.08e-08	3.24e-09
Total Materials Processing			2.64e+02	9.42e+00
Manufacturing Life-cycle Stage				
Monitor/module	Water	primary	1.08e+03	3.85e+01
LPG Production	Water, unspecified	secondary	5.00e+02	1.79e+01
Panel components	Water	primary	1.79e+02	6.38e+00
Backlight	Water	primary	1.67e+02	5.95e+00
Japanese Electric Grid	Water	model/secondary	1.49e+02	5.32e+00
Backlight	Water	primary	2.16e+01	7.70e-01
PWB Mfg.	Water	model/secondarv	1.86e+01	6.64e-01
Backlight	Water	primary	3.76e+00	1.34e-01
Fuel Oil #4 Prod.	Water, unspecified	secondary	3.68e+00	1.31e-01

Table M-2. LCD LCIA Results for the Renewable Resource Use Impact Category

Process Group	Material	LCI Data Type	Renewable Resource Use (kg)	% of Total
US electric grid	Water	model/secondary	2.20e+00	7.87e-02
LCD glass mfg.	Water	primary	1.62e+00	5.80e-02
Fuel Oil #2 Prod.	Water, unspecified	secondary	1.37e+00	4.88e-02
Fuel Oil #6 Prod.	Water, unspecified	secondary	1.21e+00	4.33e-02
Natural Gas Prod.	Water	secondary	3.34e-01	1.19e-02
Panel components	Water	primary	3.03e-01	1.08e-02
Panel components	Water	primary	2.01e-01	7.19e-03
LCD glass mfg.	Sand	primary	1.11e-01	3.97e-03
Japanese Electric Grid	Limestone	model/secondary	1.03e-01	3.68e-03
LPG Production	Limestone (CaCO3, in ground)	secondary	5.10e-02	1.82e-03
US electric grid	Limestone	model/secondary	4.66e-03	1.67e-04
Natural Gas Prod.	Limestone (CaCO3, in ground)	secondary	2.90e-03	1.04e-04
LPG Production	Sand (in ground)	secondary	1.31e-03	4.67e-05
LPG Production	Sodium chloride (NaCl, in ground or in sea)	secondary	6.00e-04	2.14e-05
Fuel Oil #4 Prod.	Limestone (CaCO3, in ground)	secondary	5.48e-04	1.96e-05
Fuel Oil #6 Prod.	Limestone (CaCO3, in ground)	secondary	2.92e-04	1.04e-05
Fuel Oil #2 Prod.	Limestone (CaCO3, in ground)	secondary	1.56e-04	5.56e-06
Fuel Oil #4 Prod.	Sand (in ground)	secondary	9.58e-06	3.42e-07
Fuel Oil #4 Prod.	Sodium chloride (NaCl, in ground or in sea)	secondary	4.39e-06	1.57e-07
Fuel Oil #2 Prod.	Sand (in ground)	secondary	3.57e-06	1.27e-07
Fuel Oil #6 Prod.	Sand (in ground)	secondary	3.14e-06	1.12e-07
Fuel Oil #2 Prod.	Sodium chloride (NaCl, in ground or in sea)	secondary	1.64e-06	5.85e-08
Fuel Oil #6 Prod.	Sodium chloride (NaCl, in ground or in sea)	secondary	1.44e-06	5.14e-08
Natural Gas Prod.	Sand (in ground)	secondary	8.68e-07	3.10e-08
Natural Gas Prod.	Sodium chloride (NaCl, in ground or in sea)	secondary	3.98e-07	1.42e-08
Total Manufacturing		·	2.13e+03	7.59e+01
Use. Maintenance and Repair Li	fe-cvcle Stage			
US electric grid	Water	model/secondary	4 25e+02	1.52e+01
US electric grid	Limestone	model/secondary	8 99e-01	3.21e-02
Total Use Maintenance and Re	nair		4.26+02	1.520+01
End of life Life avale Stage	2411		4.200+02	1.526+01
L CD landfilling	Class (in anound)		0.520.01	$2 40_{2} 02$
LCD individual	Clay (in ground)	primary	9.526-01	3.40e-02
	Cray (in ground)	secondary	7.586-01	2.046-02
LCD insingution	Sand (in ground)	primary	2.450.01	1.13e-02 8.72a.02
LCD incineration	Sand (in ground)	secondary	2.45e-01	8.75e-03
US electric grid	Water unergeified	model/secondary	8.00e-02	2.886-03
LPG Production	Limestane	secondary	4.116-02	1.4/e-03
L CD landfilling	Limestone	minouel/secondary	1.710-04	5.442.06
LCD landfilling	Sodium shlarida (NaCl. in ground or in soo)	primary	1.326-04	5.446-00
	Sodium chloride (NaCl, in ground or in sea)	primary	1.826-03	0.50e-07
LCD incineration	Limestana (CaCO2, in ground or in sea)	secondary	1.28e-05	4.59e-07
LPC Production	Send (in ground)	secondary	4.196-00	2,822,00
LPG Production	Sadium ablarida (NaCl. in ground or in soc)	secondary	1.07e-07	1.762.00
LFO FIOUUCIIOII	Sodium chloride (NaCl, in ground or in sea)	secondary	4.920-08	2.002.00
Natural Gas Prod	Sond (in ground)	secondary	-8.370-08	-2.990-09
Fuel Oil #4 Prod	Sodium chloride (NaCl in ground on in g)	secondary	-1.620-07	-0.516-09
Fuel Oil #4 Prod	Sound (in ground)	secondary	-2.02e-05	-7.230-07
Natural Gas Prod	Limostono (CoCO2, in ground)	secondary	-4.410-03	-1.586-06
Ivatural Gas Prou.	Linestone (CaCOS, in ground)	secondary	-0.096-04	-2.1/e-05

Table M-2. LCD LCIA Results for the Renewable Resource Use Impact Category

Process Group	Material	LCI Data Type	Renewable Resource Use (kg)	% of Total
Fuel Oil #4 Prod.	Limestone (CaCO3, in ground)	secondary	-2.52e-03	-9.01e-05
Natural Gas Prod.	Water	secondary	-7.02e-02	-2.51e-03
LCD incineration	Limestone (CaCO3, in ground)	secondary	-1.52e-01	-5.44e-03
LCD incineration	Water	secondary	-1.12e+00	-4.01e-02
Fuel Oil #4 Prod.	Water, unspecified	secondary	-1.69e+01	-6.05e-01
Total End-of-life			-1.59e+01	-5.68e-01
Total All Life-cycle Stages			2.80e+03	1.00e+02

Table M-2. LCD LCIA Results for the Renewable Resource Use Impact Category

Process Group	Material	LCI Data Type	Nonrenewable Resource Use (kg)	% of Total
Materials Processing Life-cycle Sta	ze	_	Resource ese (ng)	
Steel Prod., cold-rolled, semi-finished	Iron (Fe, ore)	secondary	6.65e+00	9.95e-01
Steel Prod., cold-rolled, semi-finished	Coal, average (in ground)	secondary	3.98e+00	5.96e-01
Aluminum Prod.	Bauxite (Al2O3, ore)	secondary	1.37e+00	2.05e-01
Polycarbonate Production	Natural gas (in ground)	secondary	1.29e+00	1.92e-01
Styrene-butadiene Copolymer Prod.	Petroleum (in ground)	secondary	1.03e+00	1.54e-01
Aluminum Prod.	Coal. average (in ground)	secondary	9.83e-01	1.47e-01
Lead	Lead (Pb. ore)	secondary	4.96e-01	7.42e-02
Steel Prod., cold-rolled, semi-finished	Petroleum (in ground)	secondary	4.89e-01	7.32e-02
Aluminum Prod.	Petroleum (in ground)	secondary	4.64e-01	6.95e-02
Steel Prod., cold-rolled, semi-finished	Natural gas (in ground)	secondary	4.58e-01	6.86e-02
ABS Production	Natural gas (in ground)	secondary	4 42e-01	6 62e-02
Polycarbonate Production	Petroleum (in ground)	secondary	4.34e-01	6.49e-02
Styrene-butadiene Copolymer Prod	Natural gas (in ground)	secondary	4 30e-01	6 44e-02
Steel Prod cold-rolled semi-finished	Coal lignite (in ground)	secondary	4 16e-01	6 23e-02
Polycarbonate Production	Coal average (in ground)	secondary	3 90e-01	5.85e-02
Invar	Petroleum (in ground)	secondary	3.70e-01	5.55e-02
Invar	Natural gas (in ground)	secondary	3.51e-01	5.25e-02
ABS Production	Petroleum (in ground)	secondary	2 98e-01	4.45e-02
Lead	Coal average (in ground)	secondary	2.98e-01	4.30e-02
Invar	Licon (Fe, ore)	secondary	2.536-01	3 77e 02
Polycarbonate Production	Coal lignite (in ground)	secondary	2.520-01	3.69e.02
Forycarbonate Froduction	Licen ere	secondary	2.406-01	3.096-02
Aluminum Brod	Natural cas (in ground)	secondary	2.370-01	3.330-02
Aluminum Prod	Cool lignita (in ground)	secondary	1.820.01	2,720,02
Farrita mfa	Coal, average (in ground)	secondary	1.030-01	2.730-02
L and	Natural gas (in ground)	secondary	1.746-01	2.000-02
Relycturone Prod. high impact	Natural gas (in ground)	secondary	1.30e-01	2.346-02
Forystyrene Frou, ingn-impact	Coal lignite (in ground)	secondary	1.390-01	1.020.02
Forrito mfg	Natural gas (in ground)	secondary	1.260-01	1.926-02
Petrite IIIg.	Natural gas (in ground)	secondary	1.246-01	1.856-02
A luminum Drod	Pretroleum (m ground)	secondary	1.186-01	1.700-02
Aluminum Prod.	Nislash (Nislash)	secondary	1.016-01	1.31e-02
Invar Land	Detectory (in anound)	secondary	9.806-02	1.476-02
Lead	Fetroleum (m ground)	secondary	5.74e-02	7.022.02
Lead	Durita (FaS2, and)	secondary	5.00-02	7.936-03
Lead	Pyrite (FeS2, ore)	secondary	3.096-02	7.02e-03
Ferrite mig.	Petroleum (in ground)	secondary	4.906-02	7.346-03
Invar	Tin (Sn. cm)	secondary	3.946-02	3.896-03
Steel Prod., cold-rolled, semi-finished	1 in (Sn, ore)	secondary	2.43e-02	3.64e-03
Invar	Zinc (Zn, ore)	secondary	1.92e-02	2.8/e-03
Ferrite mig.	Zinc (Zn, ore)	secondary	1.8/e-02	2.80e-03
Aluminum Prod.	Iron (Fe, ore)	secondary	1.75e-02	2.62e-03
ABS Production	l alcum (ore)	secondary	8.89e-03	1.33e-03
Steel Prod., cold-rolled, semi-finished	Raw materials (unspecified)	secondary	7.11e-03	1.06e-03
Polycarbonate Production	Suitur (S, in ground)	secondary	5.31e-03	7.95e-04
Ferrite mfg.	Pyrite (FeS2, ore)	secondary	2.19e-03	3.27e-04
ABS Production	Potassium chloride (KCl, as K2O, in ground)	secondary	1.91e-03	2.85e-04
Polycarbonate Production	Bentonite (in ground)	secondary	1.29e-03	1.94e-04
Invar	Barium sulfate (BaSO4, in ground)	secondary	1.10e-03	1.65e-04

Table M-3. CRT LCIA Results for the Nonrenewable Resource Use Impact Category

Process Group	Material	LCI Data Type	Nonrenewable Resource Use (kg)	% of Total
Polycarbonate Production	Iron (Fe, ore)	secondary	8.68e-04	1.30e-04
Aluminum Prod.	Copper (Cu, ore)	secondary	7.06e-04	1.06e-04
Aluminum Prod.	Bentonite (in ground)	secondary	6.62e-04	9.92e-05
Styrene-butadiene Copolymer Prod.	Iron (Fe, ore)	secondary	5.83e-04	8.73e-05
Invar	Raw materials (unspecified)	secondary	5.29e-04	7.91e-05
Invar	Bauxite (Al2O3, ore)	secondary	5.27e-04	7.90e-05
Ferrite mfg.	Raw materials (unspecified)	secondary	5.16e-04	7.72e-05
Styrene-butadiene Copolymer Prod.	Bauxite	secondary	4.80e-04	7.19e-05
ABS Production	Iron (Fe, ore)	secondary	3.81e-04	5.71e-05
Invar	Bentonite (in ground)	secondary	3.24e-04	4.85e-05
Steel Prod., cold-rolled, semi-finished	Pyrite (FeS2, ore)	secondary	3.04e-04	4.55e-05
Polycarbonate Production	Bauxite (Al2O3, ore)	secondary	2.77e-04	4.15e-05
ABS Production	Bauxite (Al2O3, ore)	secondary	2.54e-04	3.80e-05
Invar	Copper (Cu, ore)	secondary	1.61e-04	2.41e-05
Polystyrene Prod., high-impact	Bauxite	secondary	1.51e-04	2.26e-05
Polystyrene Prod., high-impact	Iron ore	secondary	1.27e-04	1.90e-05
ABS Production	Bentonite (in ground)	secondary	8.47e-05	1.27e-05
Styrene-butadiene Copolymer Prod.	Sulfur (S, in ground)	secondary	8.15e-05	1.22e-05
Aluminum Prod.	Lead (Pb, ore)	secondary	6.30e-05	9.43e-06
Styrene-butadiene Copolymer Prod.	Bentonite (in ground)	secondary	5.79e-05	8.67e-06
Steel Prod., cold-rolled, semi-finished	Uranium (U, ore)	secondary	4.27e-05	6.39e-06
Ferrite mfg.	Bentonite (in ground)	secondary	3.82e-05	5.72e-06
Ferrite mfg.	Bauxite (Al2O3, ore)	secondary	3.37e-05	5.05e-06
Polycarbonate Production	Uranium (U. ore)	secondary	3.13e-05	4.68e-06
Aluminum Prod.	Uranium (U, ore)	secondary	2.86e-05	4.28e-06
Steel Prod., cold-rolled, semi-finished	Bentonite (in ground)	secondary	1.79e-05	2.68e-06
Steel Prod., cold-rolled, semi-finished	Iron sulfate (FeSO4, ore)	secondary	1.63e-05	2.44e-06
Polycarbonate Production	Calcium sulfate (CaSO4, ore)	secondary	1.29e-05	1.94e-06
Ferrite mfg.	Uranium (U, ore)	secondary	1.23e-05	1.85e-06
Steel Prod., cold-rolled, semi-finished	Bauxite (Al2O3, ore)	secondary	1.18e-05	1.77e-06
Polycarbonate Production	Dolomite (in ground)	secondary	9.23e-06	1.38e-06
Styrene-butadiene Copolymer Prod.	Potassium chloride (KCl. as K2O, in ground)	secondary	8.69e-06	1.30e-06
Polycarbonate Production	Olivine ore	secondary	7.39e-06	1.11e-06
Lead	Uranium (U. ore)	secondary	6.08e-06	9.10e-07
Styrene-butadiene Copolymer Prod	Calcium sulfate (CaSO4_ore)	secondary	5.79e-06	8 67e-07
Polycarbonate Production	Potassium chloride (KCl, as K2O, in ground)	secondary	5.54e-06	8.29e-07
Invar	Iron sulfate (FeSO4, ore)	secondary	4.68e-06	7.01e-07
Ferrite mfg.	Iron sulfate (FeSO4, ore)	secondary	4.57e-06	6.84e-07
Invar	Calcium sulfate (CaSO4, ore)	secondary	4.32e-06	6.47e-07
ABS Production	Dolomite (in ground)	secondary	4 24e-06	6 34e-07
ABS Production	Olivine ore	secondary	3 39e-06	5.07e-07
Styrene-butadiene Copolymer Prod	Dolomite (in ground)	secondary	2.48e-06	3.72e-07
Polycarbonate Production	Lead (Ph. ore)	secondary	1.85e-06	2 76e-07
ABS Production	Fluorspar (CaF2 ore)	secondary	1.69e-06	2.76e 07
Styrene-butadiene Conolymer Prod	Olivine ore	secondary	1.07e-00	2.54c-07
Invar	Sulfur (S in ground)	secondary	1.62e-06	2.42e-07
Ferrite mfg	Sulfur (S, in ground)	secondary	1.52e-06	2.42e 07 2.37e 07
Invar	Boray	secondary	2 630 07	3.94e-08
Ferrite mfg	Borax	secondary	2.030-07	3.850-08
Steel Prod. cold_rolled_semi_finished	Copper (Cu. ore)	secondary	1.860.07	2 780 09
Eerrite mfg	Copper (Cu, ore)	secondary	1.000-07	2.700-08
i cirite illig.	copper (cu, orc)	secondary	1.628-07	2.730-08

Table M-3. CRT LCIA Results for the Nonrenewable Resource Use Impact Category

Tuble III 5. EKT LEIM	results for the rounderwable		Impact Catego	51 y
Process Group	Material	LCI Data Type	Nonrenewable Resource Use (kg)	% of Total
Invar	Lead (Pb, ore)	secondary	5.84e-08	8.74e-09
Steel Prod., cold-rolled, semi-finished	Lead (Pb, ore)	secondary	5.79e-08	8.67e-09
Ferrite mfg.	Lead (Pb, ore)	secondary	5.70e-08	8.53e-09
Invar	Chromium ore	secondary	3.67e-08	5.50e-09
Steel Prod., cold-rolled, semi-finished	Chromium ore	secondary	3.65e-08	5.46e-09
Ferrite mfg.	Chromium ore	secondary	3.59e-08	5.37e-09
Invar	Manganese (Mn, ore)	secondary	2.14e-08	3.20e-09
Steel Prod., cold-rolled, semi-finished	Manganese (Mn, ore)	secondary	2.13e-08	3.18e-09
Ferrite mfg.	Manganese (Mn, ore)	secondary	2.09e-08	3.13e-09
Steel Prod., cold-rolled, semi-finished	Nickel (Ni, ore)	secondary	1.24e-08	1.85e-09
Ferrite mfg.	Nickel (Ni, ore)	secondary	1.21e-08	1.82e-09
Steel Prod., cold-rolled, semi-finished	Zinc (Zn, ore)	secondary	1.35e-09	2.02e-10
Invar	Silver (Ag, ore)	secondary	9.27e-10	1.39e-10
Steel Prod., cold-rolled, semi-finished	Silver (Ag, ore)	secondary	9.20e-10	1.38e-10
Ferrite mfg.	Silver (Ag, ore)	secondary	9.05e-10	1.35e-10
Invar	Dolomite	secondary	7.26e-11	1.09e-11
Ferrite mfg.	Dolomite	secondary	7.08e-11	1.06e-11
Invar	Olivine	secondary	5.44e-11	8.15e-12
Ferrite mfg.	Olivine	secondary	5.31e-11	7.95e-12
Total Materials Processing		•	2 32e+01	3 47e+00
Manufacturing Life-cycle Stage				51170.00
LPG Production	Petroleum (in ground)	secondary	3.75e+02	5.61e+01
LPG Production	Natural gas (in ground)	secondary	4.51e+01	6.75e+00
LPG Production	Coal. average (in ground)	secondary	1.34e+01	2.01e+00
Fuel Oil #6 Prod.	Petroleum (in ground)	secondary	3.88e+00	5.81e-01
Natural Gas Prod.	Natural gas (in ground)	secondary	3.40e+00	5.09e-01
US electric grid	Coal. average (in ground)	model/secondary	2.86e+00	4.29e-01
Japanese Electric Grid	Coal, average (in ground)	model/secondary	2.28e+00	3.42e-01
Japanese Electric Grid	Petroleum (in ground)	model/secondary	1.29e+00	1.93e-01
Japanese Electric Grid	Natural gas	model/secondary	1.25e+00	1.87e-01
Fuel Oil #2 Prod.	Petroleum (in ground)	secondary	1.24e+00	1.85e-01
Glass/frit	Lead	primary	4.47e-01	6.70e-02
Fuel Oil #6 Prod	Natural gas (in ground)	secondary	2.88e-01	4 31e-02
US electric grid	Natural gas	model/secondary	2.23e-01	3.34e-02
Fuel Oil #4 Prod.	Petroleum (in ground)	secondary	1.47e-01	2.19e-02
Fuel Oil #2 Prod.	Natural gas (in ground)	secondary	1.36e-01	2.04e-02
Fuel Oil #6 Prod.	Coal. average (in ground)	secondary	1.08e-01	1.62e-02
US electric grid	Petroleum (in ground)	model/secondary	6.07e-02	9.08e-03
Glass/frit	Lead	nrimary	4 67e-02	6 99e-03
LPG Production	Bauxite (Al2O3 ore)	secondary	4 45e-02	6 66e-03
Fuel Oil #2 Prod	Coal average (in ground)	secondary	4 21e-02	6 30e-03
Natural Gas Prod	Coal average (in ground)	secondary	2 33e-02	3 49e-03
Fuel Oil #4 Prod	Natural gas (in ground)	secondary	1 34e-02	2.00e-03
Natural Gas Prod	Petroleum (in ground)	secondary	8 39e-03	1.26e-03
Glass/frit	Borax	nrimary	8 00e-03	1.200 03
Glass/frit	Silica	primary	5 330.03	7 980-04
Jananese Electric Grid	Uranium vellowcake	model/secondary	3.04e-04	4 55e-05
LPG Production	Uranium (II ore)	secondary	2.26e-04	3 380-05
End Oil #6 Prod	Bauxite (Al2O3 ore)	secondary	1 50e.04	2 25e-05
	Buunic (11205, 010)	secondary	1.500-04	2.230-03

Table M-3. CRT LCIA Results for the Nonrenewable Resource Use Impact Category

Bussess Cusur	Material	LCI Data Tuna		
Process Group		LCI Data Type	Resource Use (kg)	% of 10tal
Fuel Oil #2 Prod.	Bauxite (Al2O3, ore)	secondary	1.25e-04	1.87e-05
US electric grid	Uranium, yellowcake	model/secondary	7.74e-05	1.16e-05
Fuel Oil #4 Prod.	Bauxite (Al2O3, ore)	secondary	1.01e-05	1.52e-06
Fuel Oil #6 Prod.	Uranium (U, ore)	secondary	1.82e-06	2.73e-07
Natural Gas Prod.	Bauxite (Al2O3, ore)	secondary	8.99e-07	1.35e-07
Fuel Oil #2 Prod.	Uranium (U, ore)	secondary	7.08e-07	1.06e-07
Natural Gas Prod.	Uranium (U, ore)	secondary	4.00e-07	5.99e-08
Total Manufacturing			4.51e+02	6.75e+01
Use, Maintenance and Repair Life-	cycle Stage			
US electric grid	Coal, average (in ground)	model/secondary	1.79e+02	2.69e+01
US electric grid	Natural gas	model/secondary	1.40e+01	2.09e+00
US electric grid	Petroleum (in ground)	model/secondary	3.80e+00	5.69e-01
US electric grid	Uranium, yellowcake	model/secondary	4.85e-03	7.26e-04
Total Use, Maintenance and Repair			1.97e+02	2.95e+01
End-of-life Life-cycle Stage				
CRT landfilling	Oil (in ground)	primary	3.35e-02	5.01e-03
US electric grid	Coal, average (in ground)	model/secondary	1.79e-02	2.69e-03
CRT landfilling	Natural gas (in ground)	primary	4.52e-03	6.76e-04
LPG Production	Petroleum (in ground)	secondary	3.24e-03	4.84e-04
CRT landfilling	Coal, average (in ground)	primary	3.06e-03	4.59e-04
CRT Incineration	Iron (Fe, ore)	secondary	1.76e-03	2.64e-04
CRT landfilling	Iron (Fe, ore)	primary	1.65e-03	2.46e-04
US electric grid	Natural gas	model/secondary	1.40e-03	2.10e-04
LPG Production	Natural gas (in ground)	secondary	3.89e-04	5.82e-05
US electric grid	Petroleum (in ground)	model/secondary	3.80e-04	5.69e-05
LPG Production	Coal, average (in ground)	secondary	1.16e-04	1.73e-05
CRT landfilling	Bauxite (Al2O3, ore)	primary	8.81e-07	1.32e-07
US electric grid	Uranium, yellowcake	model/secondary	4.85e-07	7.27e-08
LPG Production	Bauxite (Al2O3, ore)	secondary	3.84e-07	5.75e-08
CRT landfilling	Uranium (U, ore)	primary	4.16e-08	6.23e-09
LPG Production	Uranium (U, ore)	secondary	1.95e-09	2.91e-10
Natural Gas Prod.	Uranium (U, ore)	secondary	-2.01e-07	-3.00e-08
Natural Gas Prod.	Bauxite (Al2O3, ore)	secondary	-4.51e-07	-6.75e-08
CRT Incineration	Bauxite (Al2O3, ore)	secondary	-6.05e-06	-9.06e-07
Fuel Oil #4 Prod.	Bauxite (Al2O3, ore)	secondary	-1.09e-04	-1.63e-05
Natural Gas Prod.	Petroleum (in ground)	secondary	-4.21e-03	-6.30e-04
Natural Gas Prod.	Coal, average (in ground)	secondary	-1.17e-02	-1.75e-03
CRT Incineration	Natural gas	secondary	-9.02e-02	-1.35e-02
Fuel Oil #4 Prod.	Natural gas (in ground)	secondary	-1.44e-01	-2.15e-02
Fuel Oil #4 Prod.	Petroleum (in ground)	secondary	-1.58e+00	-2.36e-01
Natural Gas Prod.	Natural gas (in ground)	secondary	-1.71e+00	-2.55e-01
Total End-of-life			-3 46e+00	-5 18e-01
Total All Life-cycle Stages			6.68e+02	1.00e+02

Table M-3. CRT LCIA Results for the Nonrenewable Resource Use Impact Category

Material S Processing Life-cycle Stage Natural Gas Prod. Natural (Gas Prod. Natural (Gas (prod., cold-rolled, semi-finished Coal, average (in ground) secondary 1.9.5e+00 8.94e+00 4.95e+00 4.95e+00 4.97e+00 4.97e+00 4.97e+00 4.97e+01 1.97e+01	Process Group	Material	LCI Data Type	Nonrenewable Resource Use (kg)	% of Total				
Natural Gas Prod.Natural gas (in ground)secondary2.38-e1026.52e101Steel Prod., cold-rolled, semi-finishedIron (Fe, ore)secondary1.95e+0005.86e-01Natural Gas Prod.Coal, average (in ground)secondary1.61e+004.47e-01Dylourboatte ProductionNatural Gas (in ground)secondary7.18e-011.97e-01Natural Gas Prod.Petroleum (in ground)secondary5.86e-011.161e-01Natural Gas Prod.Petroleum (in ground)secondary5.86e-011.161e-01Styrene-butadiene Copolymer Prod.Petroleum (in ground)secondary4.48e-011.21e-01PMAA Sheet Prod.Natural gas (in ground)secondary4.46e-011.15e-01Aluminum Prod.Coal, average (in ground)secondary2.42e-016.65e-02Dylourboatte ProductionPetroleum (in ground)secondary2.46e-016.16e-02Dylourboatte ProductionCoal, average (in ground)secondary2.18e-015.86e-01Steel Prod., cold-rolled, semi-finishedPetroleum (in ground)secondary2.18e-015.86e-02Steel Prod., cold-rolled, semi-finishedCoal, average (in ground)secondary2.18e-015.86e-02Steel Prod., cold-rolled, semi-finishedCoal, jignite (in ground)secondary1.38e-015.74e-02Aluminum Prod.Coal, jignite (in ground)secondary1.38e-015.74e-02PFT Resin ProductionPetroleum (in ground)secondary1.38e-045.74e-02PHT Resin Pro	Materials Processing Life-cycle Stage								
Stecl Prod., cold-rolled, semi-finished Torn (Fc, orc) secondary 1.36e+00 8.36e-01 Steel Prod., cold-rolled, semi-finished Coal, average (in ground) secondary 1.66e+00 4.47e-01 Polycarbonate Production Natural Gas (pround) secondary 5.36e-01 1.61e+00 Autural Gas Production Natural Gas (pround) secondary 5.36e-01 1.61e+00 Autural Cas Production Petroleum (in ground) secondary 4.36e-01 1.21e+01 PMMA Sheet Prod. Petroleum (in ground) secondary 4.48e-01 1.21e+01 PMMA Sheet Prod. Petroleum (in ground) secondary 2.46e-01 6.56e-02 Steel Prod., cold-rolled, semi-finished Natural gas (in ground) secondary 2.46e-01 6.56e-02 Steel Prod., cold-rolled, semi-finished Natural gas (in ground) secondary 2.14e-01 6.16e-02 Polycarbonate Production Coal, average (in ground) secondary 2.18e-01 5.96e-02 Steel Prod., cold-rolled, semi-finished Natural gas (in ground) secondary 1.38e-01 3.16e-02	Natural Gas Prod.	Natural gas (in ground)	secondary	2.38e+02	6.52e+01				
Steel Prod., cold-averlage (in ground) secondary 1.95c+00 5.36c-01 Natural Gas Prod. Coal, average (in ground) secondary 7.186-00 4.47c-01 Natural Gas Prod. Petroleum (in ground) secondary 5.86c-01 1.61c-03 Natural Gas Prod. Petroleum (in ground) secondary 5.86c-01 1.161c-01 Styme-butadiene Copolymer Prod. Petroleum (in ground) secondary 4.48c-01 1.212-01 PMAA Sheet Prod. Natural gas (in ground) secondary 4.48c-01 1.15c-01 Aluminum Prod. Coal, average (in ground) secondary 2.42c-01 6.56c-02 Steel Prod., cold-rolled, semi-finished Petroleum (in ground) secondary 2.48c-01 6.16c-02 Steel Prod., cold-rolled, semi-finished Natural gas (in ground) secondary 2.48c-01 6.16c-02 Stymen-bundlex Copolymer Prod. Coal, average (in ground) secondary 2.18c-01 5.86c-02 Stymen-bundlex Copolymer Prod. Coal, average (in ground) secondary 1.38c-01 5.86c-02 Stymen-bundlex Copolymer Prod.	Steel Prod., cold-rolled, semi-finished	Iron (Fe, ore)	secondary	3.26e+00	8.94e-01				
Natural Gas Prod. Cod., average (in ground) secondary 1.63:e-00 4.47:e-01 Polycarbonate Production Natural Qas (in ground) secondary 5.08:e-01 1.61:e01 Aluminum Prod. Bauxis (A1203, ore) secondary 5.09:e-01 1.40:e01 Sysrene-buildenc Copolymer Prod. Petroleum (in ground) secondary 4.43:e-01 1.23:e-01 PMMA Sheet Prod. Natural gas (in ground) secondary 4.43:e-01 1.23:e-01 Polycarbonate Production Petroleum (in ground) secondary 2.42:e-01 6.65:e-02 Steel Prod., cold-rolled, semi-finished Natural gas (in ground) secondary 2.24:e-01 6.16:e-02 Steel Prod., cold-rolled, semi-finished Coal, average (in ground) secondary 2.18:e-01 5.06:e-02 Steel Prod., cold-rolled, semi-finished Coal, lignite (in ground) secondary 1.38:e-01 7.74:e-02 Steel Prod., cold-rolled, semi-finished Coal, lignite (in ground) secondary 1.38:e-01 7.74:e-02 Steel Prod., cold-rolled, semi-finished Coal, lignite (in ground) secondary 1.38:e-01 <t< td=""><td>Steel Prod., cold-rolled, semi-finished</td><td>Coal, average (in ground)</td><td>secondary</td><td>1.95e+00</td><td>5.36e-01</td></t<>	Steel Prod., cold-rolled, semi-finished	Coal, average (in ground)	secondary	1.95e+00	5.36e-01				
Polycarbonate ProductionNatural gas (in ground)secondary7.18-011.97-01Natural Gas Prod.Petroleum (in ground)secondary5.86-011.61e-01Styrene-butatienc Copolymer Prod.Petroleum (in ground)secondary4.48-011.23e-01PMMA Sheet Prod.Natural gas (in ground)secondary4.48-011.23e-01Aluminum Prod.Coal, average (in ground)secondary2.42e-016.65e-02Steel Prod, cold-rolled, semi-finishedPetroleum (in ground)secondary2.42e-016.65e-02Steel Prod, cold-rolled, semi-finishedNatural gas (in ground)secondary2.42e-016.65e-02Steel Prod, cold-rolled, semi-finishedNatural gas (in ground)secondary2.48e-015.98e-02Steel Prod, cold-rolled, semi-finishedCoal, average (in ground)secondary2.18e-015.98e-02Steel Prod, cold-rolled, semi-finishedCoal, average (in ground)secondary1.78-014.77e-02Polycarbonate ProductionCoal, lignite (in ground)secondary9.30e-022.55-02Aluminum Prod.Petroleum (in ground)secondary9.36e-021.86-02Polycarbonate ProductionPetroleum (in ground)secondary9.36e-021.86-02Aluminum Prod.Coal, lignite (in ground)secondary2.37e-027.66-03Aluminum Prod.Natural gas (in ground)secondary2.37e-023.86-021.86-02PET Resin ProductionCoal, lignite (in ground)secondary2.37e-033.18e-04	Natural Gas Prod.	Coal, average (in ground)	secondary	1.63e+00	4.47e-01				
Nama Gas Prod. Petroleum (in ground) secondary 5.86-01 1.61-01 Aluminum Prod. Bauxile (AI2O3, ore) secondary 5.09-01 1.40e-01 PMMA Sheet Prod. Petroleum (in ground) secondary 4.48-01 1.22e-01 PMMA Sheet Prod. Natural gas (in ground) secondary 3.66-01 1.00e-01 Polycarborate Production Coal, average (in ground) secondary 2.42e-01 6.58e-02 Steel Prod., obt-rolled, semi-finished Petroleum (in ground) secondary 2.24e-01 6.58e-02 Steel Prod., obt-rolled, semi-finished Natural gas (in ground) secondary 2.04e-01 5.08e-02 Steel Prod., obt-rolled, semi-finished Coal, lignite (in ground) secondary 1.88e-01 5.78e-02 Steel Prod., cold-rolled, semi-finished Coal, lignite (in ground) secondary 1.38e-01 5.78e-02 Aluminum Prod. Polycarbonter Production Coal, lignite (in ground) secondary 1.38e-01 1.77e-02 Polycarbonter Production Coal, lignite (in ground) secondary 1.38e-01 1.05e-02	Polycarbonate Production	Natural gas (in ground)	secondary	7.18e-01	1.97e-01				
Aluminum Prod.Bauxite (Al203, ore)secondary5.0%-011.40e-01Stynen-butadient Copolymer Prod.Petroleum (in ground)secondary4.48e-011.23e-01PMMA Sheet Prod.Natural gas (in ground)secondary4.19e-011.15e-01Polycarbonate ProductionPetroleum (in ground)secondary2.42e-016.65e-02Steel Prod, old-rolled, semi-finishedPetroleum (in ground)secondary2.42e-016.65e-02Steel Prod, old-rolled, semi-finishedNatural gas (in ground)secondary2.42e-016.65e-02Steel Prod, old-rolled, semi-finishedCoal, average (in ground)secondary2.18e-015.66e-02Steel Prod, old-rolled, semi-finishedCoal, lignite (in ground)secondary1.48e-015.16e-02Steel Prod, old-rolled, semi-finishedCoal, lignite (in ground)secondary1.48e-015.16e-02Steel Prod, old-rolled, semi-finishedCoal, lignite (in ground)secondary1.48e-015.16e-02Aluminum Prod.Coal, lignite (in ground)secondary9.36e-022.55e-02Aluminum Prod.Coal, lignite (in ground)secondary6.86e-021.86e-02PET Resin ProductionPetroleum (in ground)secondary3.84e-021.86e-02PET Resin ProductionNatural gas (in ground)secondary2.75e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary2.75e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary2.78e-027.64e-03	Natural Gas Prod.	Petroleum (in ground)	secondary	5.86e-01	1.61e-01				
Styrene-butadiene Copolymer Prod.Petroleum (in ground)secondary4.48e-011.23e-01PMMA Sheet Prod.Natural gas (in ground)secondary4.49e-011.21e-01Aluminum Prod.Coal, average (in ground)secondary2.40e-016.65e-02Steel Prod., cold-rolled, semi-finishedPetroleum (in ground)secondary2.24e-016.65e-02Steel Prod., cold-rolled, semi-finishedNatural gas (in ground)secondary2.24e-016.16e-02Steel Prod., cold-rolled, semi-finishedCoal, lignite (in ground)secondary2.18e-015.98e-02Steel Prod., cold-rolled, semi-finishedCoal, lignite (in ground)secondary1.88e-015.16e-02Styrene-butadiene Copolymer Prod.Natural gas (in ground)secondary1.38e-013.77e-01Polycarbonate ProductionCoal, lignite (in ground)secondary9.30e-022.55e-02Aluminum Prod.Netroleum (in ground)secondary9.30e-022.55e-02Aluminum Prod.Natural gas (in ground)secondary3.84e-015.16e-02Aluminum Prod.Natural gas (in ground)secondary3.84e-021.03e-02PET Resin ProductionCoal, lignite (in ground)secondary3.75e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary3.75e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary3.75e-033.76e-03Steel Prod., cold-rolled, semi-finishedTro (Fe, 2, ore)secondary3.75e-041.86e-02 </td <td>Aluminum Prod.</td> <td>Bauxite (Al2O3, ore)</td> <td>secondary</td> <td>5.09e-01</td> <td>1.40e-01</td>	Aluminum Prod.	Bauxite (Al2O3, ore)	secondary	5.09e-01	1.40e-01				
PMMA Sheet Prod.Petroleum (in ground)secondary4.43e-011.21e-01PMMA Sheet Prod.Natural gas (in ground)secondary3.66e-011.00e-01Polycarbonate ProductionPetroleum (in ground)secondary2.42e-016.658-02Steel Prod., cold-rolled, semi-finishedPatroleum (in ground)secondary2.24e-016.58e-02Steel Prod., cold-rolled, semi-finishedNatural gas (in ground)secondary2.24e-016.58e-02Steel Prod., cold-rolled, semi-finishedCoal, average (in ground)secondary2.18e-015.98e-02Steel Prod., cold-rolled, semi-finishedCoal, lignite (in ground)secondary1.78e-014.74e-02Steel Prod., cold-rolled, semi-finishedCoal, lignite (in ground)secondary9.30e-022.18e-02Polycarbonate ProductionPetroleum (in ground)secondary9.30e-022.18e-02Aluminum Prod.Coal, lignite (in ground)secondary9.30e-022.18e-02Aluminum Prod.Coal, lignite (in ground)secondary3.76-021.03e-02Aluminum Prod.Coal, lignite (in ground)secondary2.78e-021.03e-02Aluminum Prod.Coal, lignite (in ground)secondary3.78e-021.03e-02Aluminum Prod.Coal, lignite (in ground)secondary2.78e-021.03e-02Aluminum Prod.Coal, lignite (in ground)secondary2.78e-021.38e-01PET Resin ProductionNatural gas (in ground)secondary2.78e-021.38e-04PMA Sheet	Styrene-butadiene Copolymer Prod.	Petroleum (in ground)	secondary	4.48e-01	1.23e-01				
PMMA Sheet Prod.Natural gas (in ground)secondary4.19e-011.15e-01Aluminum Prod.Coal, average (in ground)secondary3.66e-011.00e-01Olycarbonate ProductionPetroleum (in ground)secondary2.42e-016.65e-02Steel Prod., cold-rolled, semi-finishedNatural gas (in ground)secondary2.18e-016.58e-02Steel Prod., cold-rolled, semi-finishedNatural gas (in ground)secondary2.18e-015.60e-02Steel Prod., cold-rolled, semi-finishedCoal, lignic (in ground)secondary1.73e-014.74e-02Styrene-battatiene Copolymer Prod.Natural gas (in ground)secondary1.73e-014.74e-02Polyarabonate ProductionCoal, lignic (in ground)secondary9.30e-022.55e-02Aluminum Prod.Coal, lignic (in ground)secondary9.30e-022.55e-02Aluminum Prod.Coal, lignic (in ground)secondary3.76e-021.86e-02Aluminum Prod.Coal, lignic (in ground)secondary3.76e-021.05e-02Aluminum Prod.Coal, lignic (in ground)secondary2.78e-021.03e-02Aluminum Prod.Coal, lignic (in ground)secondary2.78e-023.76e-03PET Resin ProductionNatural gas (in ground)secondary2.78e-023.76e-03Steel Prod., cold-rolled, semi-finishedTin (Sn, ore)secondary3.78e-023.78e-03Steel Prod., cold-rolled, semi-finishedTin (Sn, ore)secondary3.78e-033.78e-03Steel Prod., co	PMMA Sheet Prod.	Petroleum (in ground)	secondary	4.43e-01	1.21e-01				
Aluminum Prod.Coal, average (in ground)secondary3.66e-011.00e-01Polycarbonate ProductionPetroleum (in ground)secondary2.40e-016.65e-02Steel Prod., cold-rolled, semi-finishedNatural gas (in ground)secondary2.14e-016.58e-02Steel Prod., cold-rolled, semi-finishedCoal, average (in ground)secondary2.14e-015.98e-02Steel Prod., cold-rolled, semi-finishedCoal, lignic (in ground)secondary2.14e-015.96e-02Styrene-butadiene Copolymer Prod.Natural gas (in ground)secondary1.78e-014.74e-02Polycarbonate ProductionCoal, lignite (in ground)secondary1.78e-014.74e-02Polycarbonate ProductionCoal, lignite (in ground)secondary9.30e-022.58e-02Aluminum Prod.Natural gas (in ground)secondary7.95e-022.18e-02Aluminum Prod.Coal, lignite (in ground)secondary3.84e-021.05e-02Aluminum Prod.Coal, average (in ground)secondary3.75e-031.30e-02PET Resin ProductionNatural gas (in ground)secondary2.78e-031.30e-02Aluminum Prod.Coal, average (in ground)secondary3.74e-031.84e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary3.72e-031.78e-031.79e-03MMA Sheet Prod.Sulfur (K, in ground)secondary3.72e-031.78e-031.79e-03PMMA Sheet Prod.Sulfur (K, in ground)secondary2.37e-031	PMMA Sheet Prod.	Natural gas (in ground)	secondary	4.19e-01	1.15e-01				
Polycarbonate ProductionPetroleum (in ground)secondary2.42e-016.65e-02Steel Prod., cold-rolled, semi-finishedPetroleum (in ground)secondary2.24e-016.58e-02Polycarbonate ProductionCoal, average (in ground)secondary2.18e-015.98e-02Steel Prod., cold-rolled, semi-finishedCoal, lignic (in ground)secondary2.18e-015.98e-02Streen-brudiene Copolymer Prod.Natural gas (in ground)secondary1.73e-014.74e-02Aluminum Prod.Petroleum (in ground)secondary1.73e-014.74e-02Polycarbonate ProductionCoal, lignic (in ground)secondary9.30e-022.18e-02Aluminum Prod.Natural gas (in ground)secondary7.95e-022.18e-02Aluminum Prod.Natural gas (in ground)secondary3.84e-021.08e-02PET Resin ProductionCoal, lignic (in ground)secondary3.75e-021.03e-02PET Resin ProductionCoal, ging (in ground)secondary2.78e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary2.78e-027.64e-03PMA Sheet Prod.Sulfur (S, in ground)secondary2.32e-033.77e-03Steel Prod., cold-rolled, semi-finishedTin (S, no; or)secondary2.32e-033.78e-04PMMA Sheet Prod.Sulfur (S, in ground)secondary2.97e-038.14e-04PMMA Sheet Prod.Sulfur (S, in ground)secondary2.97e-038.14e-04PMMA Sheet Prod.Bentonite (in ground	Aluminum Prod.	Coal, average (in ground)	secondary	3.66e-01	1.00e-01				
Steel Prod., cold-rolled, semi-finishedPetroleum (in ground)secondary2.40e-016.58e-02Steel Prod., cold-rolled, semi-finishedNatural gas (in ground)secondary2.18e-015.98e-02Steel Prod., cold-rolled, semi-finishedCoal, lignite (in ground)secondary2.04e-015.60e-02Styrene-butadiene Copolymer Prod.Natural gas (in ground)secondary1.73e-014.74e-02Polycarbonate ProductionCoal, lignite (in ground)secondary1.73e-014.74e-02Polycarbonate ProductionCoal, lignite (in ground)secondary9.30e-022.18e-02Aluminum Prod.Petroleum (in ground)secondary7.95e-022.18e-02Aluminum Prod.Coal, lignite (in ground)secondary3.84e-021.05e-02Aluminum Prod.Coal, lignite (in ground)secondary3.75e-021.05e-02PET Resin ProductionCoal, average (in ground)secondary3.75e-021.05e-02PET Resin ProductionNatural gas (in ground)secondary3.75e-031.07e-03Steel Prod., cold-rolled, semi-finishedTin (Sn, ore)secondary3.27e-033.14e-04PMMA Sheet Prod.Sulfur (S, in ground)secondary3.27e-033.14e-04PMMA Sheet Prod.Dolomite (in ground)secondary3.27e-033.14e-04PMMA Sheet Prod.Dolomite (in ground)secondary3.27e-033.14e-04PMMA Sheet Prod.Sulfur (S, in ground)secondary3.5ee-041.58e-04Polycarbonate Production <td>Polycarbonate Production</td> <td>Petroleum (in ground)</td> <td>secondary</td> <td>2.42e-01</td> <td>6.65e-02</td>	Polycarbonate Production	Petroleum (in ground)	secondary	2.42e-01	6.65e-02				
Steel Prod., cold-rolled, semi-finishedNatural gas (in ground)secondary2.24e-016.16e-02Polycarbonate ProductionCoal, average (in ground)secondary2.18e-015.98e-02Styrene-butadiene Copolymer Prod.Natural gas (in ground)secondary1.88e-015.16e-02Aluminum Prod.Petroleum (in ground)secondary1.38e-013.77e-02Polycarbonate ProductionCoal, lignite (in ground)secondary9.30e-022.55e-02Aluminum Prod.Natural gas (in ground)secondary9.30e-022.58e-02Aluminum Prod.Natural gas (in ground)secondary7.95e-021.88e-02Aluminum Prod.Coal, lignite (in ground)secondary3.84e-021.05e-02Aluminum Prod.Coal, average (in ground)secondary3.75e-021.05e-02PET Resin ProductionNatural gas (in ground)secondary2.78e-023.76e-03Steel Prod., cold-rolled, semi-finishedTin (Sn, ore)secondary1.23e-023.37e-03Steel Prod., cold-rolled, semi-finishedTin (Sn, ore)secondary2.97e-038.14e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.97e-038.14e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.97e-038.14e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.97e-038.14e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.07e-038.14e-04Polycarbonate ProductionBut	Steel Prod., cold-rolled, semi-finished	Petroleum (in ground)	secondary	2.40e-01	6.58e-02				
Polycarbonate ProductionCoal, average (in ground)secondary2.18e-015.98e-02Steel Prod., cold-rolled, semi-finishedCoal, lignite (in ground)secondary2.04e-015.60e-02Styrene-butadiene Copolymer Prod.Natural gas (in ground)secondary1.73e-014.74e-02Polycarbonate ProductionCoal, lignite (in ground)secondary1.73e-014.74e-02Polycarbonate ProductionPetroleum (in ground)secondary9.30e-022.55e-02Aluminum Prod.Natural gas (in ground)secondary7.95e-022.18e-02Aluminum Prod.Coal, lignite (in ground)secondary3.84e-021.86e-02Aluminum Prod.Coal, average (in ground)secondary3.75e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary3.75e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary2.78e-027.64e-03PMMA Sheet Prod.Sulfur (S, in ground)secondary1.73e-023.77e-03Steel Prod., cold-rolled, semi-finishedTin (Sn, ore)secondary2.97e-033.84e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.97e-033.84e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.97e-035.86e-04Polycarbonate ProductionBentonite (in ground)secondary2.97e-035.86e-04Polycarbonate ProductionBentonite (in ground)secondary2.97e-035.86e-04PMMA Sheet Prod.Raw mate	Steel Prod., cold-rolled, semi-finished	Natural gas (in ground)	secondary	2.24e-01	6.16e-02				
Steel Prod., cold-rolled, semi-finishedCoal, lignie (in ground)secondary2.04e-015.66e-02Styrene-butadiene Copolymer Prod.Natural gas (in ground)secondary1.73e-014.74e-02Polycarbonate ProductionCoal, lignite (in ground)secondary1.73e-014.74e-02PET Resin ProductionPetroleum (in ground)secondary9.30e-022.55e-02Aluminum Prod.Coal, lignite (in ground)secondary7.95e-022.18e-02Aluminum Prod.Coal, lignite (in ground)secondary3.84e-021.05e-02PET Resin ProductionNatural gas (in ground)secondary3.77e-021.03e-02Aluminum Prod.Coal, lignite (in ground)secondary3.78e-021.03e-02Aluminum Prod.Pyrite (FeS2, ore)secondary2.78e-027.64e-03PMMA Sheet Prod.Sulfur (S, in ground)secondary1.23e-023.37e-03Steel Prod., cold-rolled, semi-finishedTin (fs, ore)secondary6.32e-031.79e-03Steel Prod., cold-rolled, semi-finishedRaw materials (unspecified)secondary2.97e-038.14e-04PMMA Sheet Prod.Dolomite (in ground)secondary2.97e-038.14e-04PMMA Sheet Prod.Dolomite (in ground)secondary3.56e-04Polycarbonate ProductionButtorite (in ground)secondary3.57e-041.88e-04PMMA Sheet Prod.Rotonite (in ground)secondary3.57e-041.88e-04PMMA Sheet Prod.Rav materials (unspecified)secondary <td>Polycarbonate Production</td> <td>Coal, average (in ground)</td> <td>secondary</td> <td>2.18e-01</td> <td>5.98e-02</td>	Polycarbonate Production	Coal, average (in ground)	secondary	2.18e-01	5.98e-02				
Styrene-butadiene Copolymer Prod.Natural gas (in ground)secondary1.88e-015.16e-02Aluminum Prod.Petroleum (in ground)secondary1.73e-014.74e-02Polycarbonate ProductionCoal, lignite (in ground)secondary9.30e-022.55e-02Aluminum Prod.Natural gas (in ground)secondary9.30e-022.18e-02Aluminum Prod.Coal, average (in ground)secondary6.80e-021.86e-02Aluminum Prod.Coal, average (in ground)secondary3.84e-021.05e-02Aluminum Prod.Pyrite (FeS2, ore)secondary3.75e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary2.78e-027.64e-03Steel Prod.Sulfur (S, in ground)secondary2.78e-027.64e-03Steel Prod.Sulfur (S, in ground)secondary3.27e-033.7e-03Steel Prod.Cold-rolled, semi-finishedTin (Sn, ore)secondary3.49e-039.56e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.97e-035.68e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.97e-035.68e-04Polycarbonate ProductionBentonite (in ground)secondary7.22e-041.98e-04PMMA Sheet Prod.Bourite (in ground)secondary7.22e-041.98e-04PMMA Sheet Prod.Bourite (in ground)secondary7.22e-041.98e-04PMMA Sheet Prod.Bourite (in ground)secondary7.22e-041.98e-04 <tr< td=""><td>Steel Prod., cold-rolled, semi-finished</td><td>Coal, lignite (in ground)</td><td>secondary</td><td>2.04e-01</td><td>5.60e-02</td></tr<>	Steel Prod., cold-rolled, semi-finished	Coal, lignite (in ground)	secondary	2.04e-01	5.60e-02				
Aluminum Prod.Petroleum (in ground)secondary1.73e-014.74e-02Polycarbonate ProductionCoal, lignite (in ground)secondary9.30e-022.55e-02Aluminum Prod.Natural gas (in ground)secondary9.30e-022.55e-02Aluminum Prod.Coal, lignite (in ground)secondary6.80e-021.86e-02Aluminum Prod.Coal, lignite (in ground)secondary3.84e-021.05e-02Aluminum Prod.Pyrite (FeS2, ore)secondary3.75e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary2.78e-027.54e-02Aluminum Prod.Sulfur (S, in ground)secondary1.23e-023.37e-03PMAA Sheet Prod.Sulfur (S, in ground)secondary1.19e-023.27e-03Aluminum Prod.Iron (Fe, ore)secondary1.19e-023.27e-03Aluminum Prod.Iron (Fe, ore)secondary3.49e-039.56e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.97e-038.14e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.97e-038.14e-04Polycarbonate ProductionBentonite (in ground)secondary2.97e-038.14e-04PMMA Sheet Prod.Dolomite (in ground)secondary3.78e-041.38e-04Polycarbonate ProductionBentonite (in ground)secondary3.78e-041.38e-04PMMA Sheet Prod.Iron (Fe, ore)secondary3.78e-041.38e-04PMMA Sheet Prod.Iron (Fe, ore)	Styrene-butadiene Copolymer Prod.	Natural gas (in ground)	secondary	1.88e-01	5.16e-02				
Polycarbonate ProductionCoal, lignife (in ground)secondary1.38e-013.77e-02PET Resin ProductionPetroleum (in ground)secondary9.30e-022.55e-02Aluminum Prod.Coal, lignife (in ground)secondary7.95e-022.18e-02Aluminum Prod.Coal, lignife (in ground)secondary3.84e-021.86e-02PET Resin ProductionCoal, average (in ground)secondary3.75e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary2.78e-027.64e-03PMMA Sheet Prod.Sulfur (S, in ground)secondary1.23e-023.77e-03Steel Prod., cold-rolled, semi-finishedTin (Sn, ore)secondary6.35e-031.79e-03Steel Prod., cold-rolled, semi-finishedRaw materials (unspecified)secondary2.97e-038.14e-04POlycarbonate ProductionSulfur (S, in ground)secondary2.97e-038.14e-04PMMA Sheet Prod.Dolomite (in ground)secondary2.97e-038.14e-04PMMA Sheet Prod.Bentonite (in ground)secondary3.53e-041.38e-04Polycarbonate ProductionIron (Fe, ore)secondary3.53e-049.86e-05PMMA Sheet Prod.Iron (Fe, ore)secondary3.57e-041.88e-04Polycarbonate ProductionIron (Fe, ore)secondary3.57e-041.58e-04Polycarbonate ProductionIron (Fe, ore)secondary3.57e-041.58e-04PMMA Sheet Prod.Iron (Fe, ore)secondary2.57e-041.58e-04 </td <td>Aluminum Prod.</td> <td>Petroleum (in ground)</td> <td>secondary</td> <td>1.73e-01</td> <td>4.74e-02</td>	Aluminum Prod.	Petroleum (in ground)	secondary	1.73e-01	4.74e-02				
PET Resin ProductionPetroleum (in ground)secondary9.30e-022.55e-02Aluminum Prod.Natural gas (in ground)secondary7.95e-022.18e-02Aluminum Prod.Coal, lignite (in ground)secondary3.84e-021.05e-02Aluminum Prod.Pyrite (FeS2, ore)secondary3.75e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary3.75e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary2.78e-027.64e-03PMMA Sheet Prod.Sulfur (S, in ground)secondary1.19e-023.27e-03Steel Prod., cold-rolled, semi-finishedTin (Sn, ore)secondary1.9e-023.27e-03Steel Prod., cold-rolled, semi-finishedRaw materials (unspecified)secondary2.07e-039.56e-04PMMA Sheet Prod.Dolomite (in ground)secondary2.07e-035.68e-04PMMA Sheet Prod.Dolomite (in ground)secondary7.22e-041.98e-04PMMA Sheet Prod.Raw materials (unspecified)secondary3.57e-041.58e-04PMMA Sheet Prod.Raw materials (unspecified)secondary3.53e-049.68e-05PMMA Sheet Prod.Iron (Fe, ore)secondary3.53e-047.22e-04PMMA Sheet Prod.Iron (Fe, ore)secondary3.53e-047.22e-04PMMA Sheet Prod.Iron (Fe, ore)secondary3.53e-047.22e-05Aluminum Prod.Copper (Cu, ore)secondary3.53e-047.21e-05Styrene-butadiene Co	Polycarbonate Production	Coal, lignite (in ground)	secondary	1.38e-01	3.77e-02				
Aluminum Prod.Natural gs (in ground)secondary7.95e-022.18e-02Aluminum Prod.Coal, lignite (in ground)secondary6.80e-021.86e-02PET Resin ProductionCoal, average (in ground)secondary3.84e-021.05e-02PET Resin ProductionNatural gas (in ground)secondary3.75e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary2.78e-027.64e-03PMMA Sheet Prod.Suffur (S, in ground)secondary1.23e-023.37e-03Steel Prod., cold-rolled, semi-finishedTin (Sn, ore)secondary6.53e-031.79e-03Steel Prod., cold-rolled, semi-finishedRaw materials (unspecified)secondary2.97e-038.14e-04Polycarbonate ProductionSuffur (S, in ground)secondary2.07e-035.68e-04Polycarbonate ProductionBentonic (in ground)secondary7.22e-041.98e-04Polycarbonate ProductionIron (Fe, ore)secondary3.37e-048.42e-05PMMA Sheet Prod.Iron (Fe, ore)secondary3.37e-048.42e-05PMMA Sheet Prod.Iron (Fe, ore)secondary3.37e-048.42e-04PMMA Sheet Prod.Iron (Fe, ore)secondary3.37e-048.42e-05PMMA Sheet Prod.Iron (Fe, ore)secondary3.37e-048.42e-05Aluminum Prod.Iron (Fe, ore)secondary3.37e-048.42e-05Aluminum Prod.Iron (Fe, ore)secondary3.37e-048.42e-05Aluminum Prod.	PET Resin Production	Petroleum (in ground)	secondary	9.30e-02	2.55e-02				
Aluminum Prod.Coal, lignite (in ground)secondary6.80e-021.86e-02PET Resin ProductionCoal, average (in ground)secondary3.84e-021.05e-02Aluminum Prod.Pyrite (FeS2, ore)secondary3.75e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary2.78e-027.64e-03PMMA Sheet Prod.Sulfur (S, in ground)secondary1.23e-023.37e-03Steel Prod., cold-rolled, semi-finishedTin (Sn, ore)secondary6.53e-031.79e-03Steel Prod., cold-rolled, semi-finishedRaw materials (unspecified)secondary2.97e-038.14e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.07e-035.68e-04Polycarbonate ProductionBentonite (in ground)secondary2.07e-035.68e-04Polycarbonate ProductionBentonite (in ground)secondary7.22e-041.98e-04PMMA Sheet Prod.Raw materials (unspecified)secondary3.53e-049.68e-05Polycarbonate ProductionIron (Fe, ore)secondary3.53e-049.68e-05PMMA Sheet Prod.Raw materials (unspecified)secondary3.55e-041.33e-04PMMA Sheet Prod.Iron (Fe, ore)secondary3.55e-047.06e-05PMMA Sheet Prod.Iron (Fe, ore)secondary3.55e-047.06e-05PMMA Sheet Prod.Bauxite (Al203, ore)secondary2.55e-047.06e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.16e-04<	Aluminum Prod.	Natural gas (in ground)	secondary	7.95e-02	2.18e-02				
PET Resin ProductionCoal, average (in ground)secondary3.84e-021.05e-02Aluminum Prod.Pyrite (FeS2, ore)secondary3.75e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary2.78e-027.64e-03PMA Sheet Prod.Sulfur (S, in ground)secondary1.23e-023.37e-03Steel Prod., cold-rolled, semi-finishedTin (Sn, ore)secondary1.9e-023.27e-03Aluminum Prod.Iron (Fe, ore)secondary6.53e-031.79e-03Steel Prod., cold-rolled, semi-finishedRaw materials (unspecified)secondary2.97e-038.14e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.07e-035.68e-04Polycarbonate ProductionBentonite (in ground)secondary7.22e-041.58e-04Polycarbonate ProductionBentonite (in ground)secondary7.25e-041.58e-04Polycarbonate ProductionIron (Fe, ore)secondary3.53e-049.68e-05PMMA Sheet Prod.Raw materials (unspecified)secondary3.07e-048.42e-05Aluminum Prod.Bauxite (Al2O3, ore)secondary2.55e-047.21e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.55e-047.21e-05Styrene-butadiene Copolymer Prod.Bauxitesecondary2.16e-045.76e-05Polycarbonate ProductionBentonite (in ground)secondary2.10e-045.76e-05Styrene-butadiene Copolymer Prod.Bauxitesecondary2.12e-0	Aluminum Prod.	Coal, lignite (in ground)	secondary	6.80e-02	1.86e-02				
Aluminum Prod.Pyrite (FeS2, ore)secondary3.75e-021.03e-02PET Resin ProductionNatural gas (in ground)secondary2.78e-027.64e-03PMMA Sheet Prod.Sulfur (S, in ground)secondary1.23e-023.37e-03Steel Prod., cold-rolled, semi-finishedTin (Sn, ore)secondary1.23e-023.27e-03Aluminum Prod.Iron (Fe, ore)secondary6.53e-031.79e-03Steel Prod., cold-rolled, semi-finishedRaw materials (unspecified)secondary2.97e-038.14e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.07e-035.68e-04Polycarbonate ProductionBentonite (in ground)secondary7.22e-041.98e-04PMMA Sheet Prod.Dolomite (in ground)secondary5.75e-041.58e-04Polycarbonate ProductionIron (Fe, ore)secondary3.07e-048.42e-05PMMA Sheet Prod.Iron (Fe, ore)secondary3.07e-048.42e-05Aluminum Prod.Bauxite (Al2O3, ore)secondary2.63e-047.21e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.55e-047.00e-05Aluminum Prod.Bentonite (in ground)secondary2.47e-046.77e-05Aluminum Prod.Bentonite (in ground)secondary2.63e-047.21e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.63e-047.21e-05Aluminum Prod.Bentonite (in ground)secondary2.47e-046.77e-05St	PET Resin Production	Coal, average (in ground)	secondary	3.84e-02	1.05e-02				
PET Resin ProductionNatural gas (in ground)secondary2.78e-027.64e-03PMMA Sheet Prod.Sulfur (S, in ground)secondary1.23e-023.37e-03Steel Prod., cold-rolled, semi-finishedTin (Sn, ore)secondary1.19e-023.27e-03Aluminum Prod.Iron (Fe, ore)secondary6.53e-031.79e-03Steel Prod., cold-rolled, semi-finishedRaw materials (unspecified)secondary2.97e-038.14e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.07e-035.68e-04Polycarbonate ProductionBentonite (in ground)secondary2.07e-035.68e-04Polycarbonate ProductionBentonite (in ground)secondary2.75e-041.58e-04Polycarbonate ProductionIron (Fe, ore)secondary3.37e-049.68e-05PMMA Sheet Prod.Iron (Fe, ore)secondary3.07e-048.42e-05Aluminum Prod.Copper (Cu, ore)secondary3.07e-048.42e-05Aluminum Prod.Copper (Cu, ore)secondary2.55e-047.00e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.47e-046.77e-05Styrene-butadiene Copolymer Prod.Bauxite (Al203, ore)secondary2.10e-045.76e-05Polycarbonate ProductionBauxite (Al203, ore)secondary2.10e-044.09e-05Styrene-butadiene Copolymer Prod.Bauxite (Al203, ore)secondary1.21e-043.32e-05Polycarbonate ProductionBauxite (Al203, ore)secondary	Aluminum Prod.	Pyrite (FeS2, ore)	secondary	3.75e-02	1.03e-02				
PMMA Sheet Prod.Sulfur (S, in ground)secondary1.23e-023.37e-03Steel Prod., cold-rolled, semi-finishedTin (Sn, ore)secondary1.19e-023.27e-03Aluminum Prod.Iron (Fe, ore)secondary6.53e-031.79e-03Steel Prod., cold-rolled, semi-finishedRaw materials (unspecified)secondary2.97e-038.14e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.97e-038.14e-04PMMA Sheet Prod.Dolomite (in ground)secondary2.07e-035.68e-04Polycarbonate ProductionBentonite (in ground)secondary7.22e-041.98e-04PMMA Sheet Prod.Raw materials (unspecified)secondary5.75e-041.58e-04Polycarbonate ProductionIron (Fe, ore)secondary4.85e-041.33e-04PMMA Sheet Prod.Iron (Fe, ore)secondary3.07e-048.42e-05PMMA Sheet Prod.Iron (Fe, ore)secondary3.07e-048.42e-05Aluminum Prod.Copper (Cu, ore)secondary2.63e-047.21e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.55e-047.00e-05Aluminum Prod.Bentonite (in ground)secondary2.10e-045.76e-05Polycarbonate ProductionBauxite (Al2O3, ore)secondary2.10e-045.76e-05Styrene-butadiene Copolymer Prod.Bauxite (Al2O3, ore)secondary1.21e-043.32e-05Steel Prod., cold-rolled, semi-finishedPyrite (FeS2, ore)secondary1.21e-04<	PET Resin Production	Natural gas (in ground)	secondary	2.78e-02	7.64e-03				
Steel Prod., cold-rolled, semi-finishedTin (Sn, ore)secondary1.19e-023.27e-03Aluminum Prod.Iron (Fe, ore)secondary6.53e-031.79e-03Steel Prod., cold-rolled, semi-finishedRaw materials (unspecified)secondary2.97e-038.14e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.97e-038.14e-04Polycarbonate ProductionBentonite (in ground)secondary2.07e-035.68e-04Polycarbonate ProductionBentonite (in ground)secondary7.22e-041.98e-04PMMA Sheet Prod.Raw materials (unspecified)secondary4.85e-041.33e-04PMMA Sheet Prod.Iron (Fe, ore)secondary3.53e-049.68e-05PMMA Sheet Prod.Iron (Fe, ore)secondary3.07e-048.42e-05Aluminum Prod.Copper (Cu, ore)secondary2.63e-047.21e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.55e-047.0e-05Aluminum Prod.Copper (Cu, ore)secondary2.10e-046.77e-05Styrene-butadiene Copolymer Prod.Bauxite (Al2O3, ore)secondary2.10e-044.09e-05Pet Resin ProductionBauxite (Al2O3, ore)secondary1.21e-043.32e-05Natural Gas Prod.Bauxite (Al2O3, ore)secondary1.21e-043.32e-05Styrene-butadiene Copolymer Prod.Bauxite (Al2O3, ore)secondary1.21e-043.32e-05Polycarbonate ProductionBauxite (Al2O3, ore)secondary1.21e-04	PMMA Sheet Prod.	Sulfur (S. in ground)	secondary	1.23e-02	3.37e-03				
Aluminum Prod.Iron (Fe, ore)secondary6.53e-031.79e-03Steel Prod., cold-rolled, semi-finishedRaw materials (unspecified)secondary2.97e-038.14e-04PMMA Sheet Prod.Dolomite (in ground)secondary2.07e-035.68e-04Polycarbonate ProductionBentonite (in ground)secondary7.22e-041.98e-04PMMA Sheet Prod.Raw materials (unspecified)secondary5.75e-041.58e-04Polycarbonate ProductionIron (Fe, ore)secondary4.85e-041.33e-04Polycarbonate ProductionIron (Fe, ore)secondary3.53e-049.68e-05PMMA Sheet Prod.Iron (Fe, ore)secondary3.07e-048.42e-05PMMA Sheet Prod.Bauxite (Al2O3, ore)secondary2.65e-047.21e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.55e-047.00e-05Aluminum Prod.Copper (Cu, ore)secondary2.47e-046.77e-05Styrene-butadiene Copolymer Prod.Bauxitesecondary2.10e-045.76e-05Polycarbonate ProductionBauxite (Al2O3, ore)secondary1.21e-043.32e-05Steel Prod., cold-rolled, semi-finishedPyrite (FeS2, ore)secondary1.21e-043.32e-05Natural Gas Prod.Bauxite (Al2O3, ore)secondary1.21e-043.32e-05Pet Resin ProductionBauxite (Al2O3, ore)secondary1.21e-043.32e-05Polycarbonate ProductionBauxite (Al2O3, ore)secondary1.21e-043.32e-05<	Steel Prod., cold-rolled, semi-finished	Tin (Sn. ore)	secondary	1.19e-02	3.27e-03				
Steel Prod., cold-rolled, semi-finishedRaw materials (unspecified)secondary3.49e-039.56e-04Polycarbonate ProductionSulfur (S, in ground)secondary2.97e-038.14e-04PMMA Sheet Prod.Dolomite (in ground)secondary2.07e-035.68e-04Polycarbonate ProductionBentonite (in ground)secondary7.22e-041.98e-04PMMA Sheet Prod.Raw materials (unspecified)secondary5.75e-041.58e-04Polycarbonate ProductionIron (Fe, ore)secondary3.53e-049.68e-05PMMA Sheet Prod.Iron (Fe, ore)secondary3.07e-048.42e-05Aluminum Prod.Bauxite (Al2O3, ore)secondary2.63e-047.21e-05Aluminum Prod.Bentonite (in ground)secondary2.55e-047.00e-05Aluminum Prod.Bentonite (in ground)secondary2.47e-046.77e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.10e-045.76e-05Polycarbonate ProductionBauxite (Al2O3, ore)secondary1.55e-044.25e-05Steel Prod., cold-rolled, semi-finishedPyrite (FeS2, ore)secondary1.55e-044.25e-05Steel Prod.Coal, Lignite (in ground)secondary3.26e-059.78e-06Petr Resin ProductionBauxite (Al2O3, ore)secondary3.56e-059.78e-06Petr Resin ProductionBauxite (Al2O3, ore)secondary3.56e-059.78e-06Natural Gas Prod.Bauxite (Al2O3, ore)secondary3.56e-059.7	Aluminum Prod.	Iron (Fe. ore)	secondary	6.53e-03	1.79e-03				
Polycarbonate ProductionSulfur (S, in ground)secondary2.97e-038.14e-04PMMA Sheet Prod.Dolomite (in ground)secondary2.07e-035.68e-04Polycarbonate ProductionBentonite (in ground)secondary7.22e-041.98e-04PMMA Sheet Prod.Raw materials (unspecified)secondary5.75e-041.58e-04Polycarbonate ProductionIron (Fe, ore)secondary4.85e-041.33e-04PMMA Sheet Prod.Iron (Fe, ore)secondary3.53e-049.68e-05PMMA Sheet Prod.Bauxite (Al2O3, ore)secondary3.07e-048.42e-05Aluminum Prod.Copper (Cu, ore)secondary2.63e-047.21e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.55e-047.00e-05Aluminum Prod.Bentonite (in ground)secondary2.47e-046.77e-05Styrene-butadiene Copolymer Prod.Bauxitesecondary2.10e-045.76e-05Polycarbonate ProductionBauxite (Al2O3, ore)secondary1.55e-044.25e-05Steel Prod., cold-rolled, semi-finishedPyrite (FeS2, ore)secondary1.21e-043.32e-05PET Resin ProductionCoal, lignite (in ground)secondary3.56e-051.72e-05Styrene-butadiene Copolymer Prod.Bauxite (Al2O3, ore)secondary1.21e-043.32e-05PET Resin ProductionCoal, lignite (in ground)secondary3.56e-051.72e-05Styrene-butadiene Copolymer Prod.Sulfur (S, in ground)secondary3.56e-0	Steel Prod., cold-rolled, semi-finished	Raw materials (unspecified)	secondary	3.49e-03	9.56e-04				
PMMA Sheet Prod.Dolomite (in ground)secondary2.07e-035.68e-04Polycarbonate ProductionBentonite (in ground)secondary7.22e-041.98e-04PMMA Sheet Prod.Raw materials (unspecified)secondary5.75e-041.58e-04Polycarbonate ProductionIron (Fe, ore)secondary4.85e-041.33e-04PMMA Sheet Prod.Iron (Fe, ore)secondary3.53e-049.68e-05PMMA Sheet Prod.Bauxite (Al2O3, ore)secondary3.07e-048.42e-05Aluminum Prod.Copper (Cu, ore)secondary2.63e-047.21e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.55e-047.00e-05Aluminum Prod.Bentonite (in ground)secondary2.47e-046.77e-05Styrene-butadiene Copolymer Prod.Bauxitesecondary2.10e-045.76e-05Polycarbonate ProductionBauxite (Al2O3, ore)secondary1.55e-044.25e-05Steel Prod., cold-rolled, semi-finishedPyrite (FeS2, ore)secondary1.55e-044.25e-05PET Resin ProductionCoal, lignite (in ground)secondary1.21e-043.32e-05Styrene-butadiene Copolymer Prod.Bauxite (Al2O3, ore)secondary1.21e-043.32e-05PET Resin ProductionCoal, lignite (in ground)secondary1.21e-043.32e-05Styrene-butadiene Copolymer Prod.Sulfur (S, in ground)secondary2.90e-051.72e-05PET Resin ProductionBauxite (Al2O3, ore)secondary2.90e-05 <td>Polycarbonate Production</td> <td>Sulfur (S. in ground)</td> <td>secondary</td> <td>2.97e-03</td> <td>8.14e-04</td>	Polycarbonate Production	Sulfur (S. in ground)	secondary	2.97e-03	8.14e-04				
Polycarbonate ProductionBentonite (in ground)secondary7.22e-041.98e-04PMMA Sheet Prod.Raw materials (unspecified)secondary5.75e-041.58e-04Polycarbonate ProductionIron (Fe, ore)secondary4.85e-041.33e-04PMMA Sheet Prod.Iron (Fe, ore)secondary3.53e-049.68e-05PMMA Sheet Prod.Bauxite (Al2O3, ore)secondary3.07e-048.42e-05Aluminum Prod.Copper (Cu, ore)secondary2.63e-047.21e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.55e-047.00e-05Aluminum Prod.Bentonite (in ground)secondary2.47e-046.77e-05Styrene-butadiene Copolymer Prod.Bauxitesecondary2.10e-045.76e-05Polycarbonate ProductionBauxite (Al2O3, ore)secondary1.55e-044.25e-05Steel Prod., cold-rolled, semi-finishedPyrite (FeS2, ore)secondary1.21e-043.32e-05Natural Gas Prod.Bauxite (Al2O3, ore)secondary1.21e-043.32e-05Natural Gas Prod.Bauxite (Al2O3, ore)secondary3.56e-059.78e-06PET Resin ProductionCoal, lignite (in ground)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary2.90e-057.97e-06Nat	PMMA Sheet Prod.	Dolomite (in ground)	secondary	2.07e-03	5.68e-04				
PMMA Sheet Prod.Raw materials (unspecified)secondary5.75e-041.58e-04Polycarbonate ProductionIron (Fe, ore)secondary4.85e-041.33e-04PMMA Sheet Prod.Iron (Fe, ore)secondary3.53e-049.68e-05PMMA Sheet Prod.Bauxite (Al2O3, ore)secondary3.07e-048.42e-05Aluminum Prod.Copper (Cu, ore)secondary2.63e-047.21e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.55e-047.00e-05Aluminum Prod.Bentonite (in ground)secondary2.47e-046.77e-05Styrene-butadiene Copolymer Prod.Bauxitesecondary2.10e-045.76e-05Polycarbonate ProductionBauxite (Al2O3, ore)secondary1.55e-044.25e-05Steel Prod., cold-rolled, semi-finishedPyrite (FeS2, ore)secondary1.49e-044.09e-05PET Resin ProductionCoal, lignite (in ground)secondary1.21e-043.32e-05Natural Gas Prod.Bauxite (Al2O3, ore)secondary3.56e-051.72e-05Styrene-butadiene Copolymer Prod.Sulfur (S, in ground)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary2.09e-057.97e-06Natural Gas Prod.Uranium (U, ore)secondary2.63e-057.22e-06PET Resin ProductionIron (Fe, ore)secondary2.63e-057.22e-06Natu	Polycarbonate Production	Bentonite (in ground)	secondary	7.22e-04	1.98e-04				
Polycarbonate ProductionIron (Fe, ore)secondary4.85e-041.33e-04PMMA Sheet Prod.Iron (Fe, ore)secondary3.53e-049.68e-05PMMA Sheet Prod.Bauxite (Al2O3, ore)secondary3.07e-048.42e-05Aluminum Prod.Copper (Cu, ore)secondary2.63e-047.21e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.55e-047.00e-05Aluminum Prod.Bentonite (in ground)secondary2.47e-046.77e-05Styrene-butadiene Copolymer Prod.Bauxitesecondary2.10e-045.76e-05Polycarbonate ProductionBauxite (Al2O3, ore)secondary1.55e-044.25e-05Steel Prod., cold-rolled, semi-finishedPyrite (FeS2, ore)secondary1.21e-043.32e-05Natural Gas Prod.Bauxite (Al2O3, ore)secondary1.21e-043.32e-05Styrene-butadiene Copolymer Prod.Bauxite (Al2O3, ore)secondary1.21e-043.32e-05PET Resin ProductionCoal, lignite (in ground)secondary1.21e-043.32e-05Natural Gas Prod.Bauxite (Al2O3, ore)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary2.90e-057.97e-06Natural Gas Prod.Uranium (U, ore)secondary2.79e-057.67e-06PET Resin ProductionIron (Fe, ore)secondary2.63e-057.22e-06	PMMA Sheet Prod.	Raw materials (unspecified)	secondary	5.75e-04	1.58e-04				
PMMA Sheet Prod.Iron (Fe, ore)secondary3.53e-049.68e-05PMMA Sheet Prod.Bauxite (Al2O3, ore)secondary3.07e-048.42e-05Aluminum Prod.Copper (Cu, ore)secondary2.63e-047.21e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.55e-047.00e-05Aluminum Prod.Bentonite (in ground)secondary2.47e-046.77e-05Styrene-butadiene Copolymer Prod.Bauxitesecondary2.10e-045.76e-05Polycarbonate ProductionBauxite (Al2O3, ore)secondary1.55e-044.25e-05Steel Prod., cold-rolled, semi-finishedPyrite (FeS2, ore)secondary1.21e-043.32e-05Natural Gas Prod.Bauxite (Al2O3, ore)secondary1.21e-043.32e-05Styrene-butadiene Copolymer Prod.Sulfur (S, in ground)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary2.90e-057.97e-06Natural Gas Prod.Uranium (U, ore)secondary2.79e-057.67e-06PET Resin ProductionIron (Fe, ore)secondary2.79e-057.67e-06PET Resin ProductionIron (Fe, ore)secondary2.79e-057.67e-06Natural Gas Prod	Polycarbonate Production	Iron (Fe, ore)	secondary	4.85e-04	1.33e-04				
PMMA Sheet Prod.Bauxite (Al2O3, ore)secondary3.07e-048.42e-05Aluminum Prod.Copper (Cu, ore)secondary2.63e-047.21e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.55e-047.00e-05Aluminum Prod.Bentonite (in ground)secondary2.47e-046.77e-05Styrene-butadiene Copolymer Prod.Bauxitesecondary2.10e-045.76e-05Polycarbonate Copolymer Prod.Bauxite (Al2O3, ore)secondary1.55e-044.25e-05Steel Prod., cold-rolled, semi-finishedPyrite (FeS2, ore)secondary1.49e-044.09e-05PET Resin ProductionCoal, lignite (in ground)secondary1.21e-043.32e-05Natural Gas Prod.Bauxite (Al2O3, ore)secondary5.76e-051.72e-05Styrene-butadiene Copolymer Prod.Sulfur (S, in ground)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary2.90e-057.97e-06Natural Gas Prod.Uranium (U, ore)secondary2.79e-057.67e-06PET Resin ProductionIron (Fe, ore)secondary2.63e-057.22e-06	PMMA Sheet Prod.	Iron (Fe, ore)	secondary	3.53e-04	9.68e-05				
Aluminum Prod.Copper (Cu, ore)secondary2.63e-047.21e-05Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.55e-047.00e-05Aluminum Prod.Bentonite (in ground)secondary2.47e-046.77e-05Styrene-butadiene Copolymer Prod.Bauxitesecondary2.10e-045.76e-05Polycarbonate ProductionBauxite (Al2O3, ore)secondary1.55e-044.25e-05Steel Prod., cold-rolled, semi-finishedPyrite (FeS2, ore)secondary1.49e-044.09e-05PET Resin ProductionCoal, lignite (in ground)secondary1.21e-043.32e-05Natural Gas Prod.Bauxite (Al2O3, ore)secondary6.28e-051.72e-05Styrene-butadiene Copolymer Prod.Sulfur (S, in ground)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary2.90e-057.97e-06Natural Gas Prod.Uranium (U, ore)secondary2.79e-057.67e-06PET Resin ProductionIron (Fe, ore)secondary2.79e-057.27e-05	PMMA Sheet Prod.	Bauxite (Al2O3, ore)	secondary	3.07e-04	8.42e-05				
Styrene-butadiene Copolymer Prod.Iron (Fe, ore)secondary2.55e-047.00e-05Aluminum Prod.Bentonite (in ground)secondary2.47e-046.77e-05Styrene-butadiene Copolymer Prod.Bauxitesecondary2.10e-045.76e-05Polycarbonate ProductionBauxite (Al2O3, ore)secondary1.55e-044.25e-05Steel Prod., cold-rolled, semi-finishedPyrite (FeS2, ore)secondary1.49e-044.09e-05PET Resin ProductionCoal, lignite (in ground)secondary1.21e-043.32e-05Natural Gas Prod.Bauxite (Al2O3, ore)secondary6.28e-051.72e-05Styrene-butadiene Copolymer Prod.Sulfur (S, in ground)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary2.90e-057.97e-06Natural Gas Prod.Uranium (U, ore)secondary2.79e-057.67e-06PET Resin ProductionIron (Fe, ore)secondary2.79e-057.22e-06	Aluminum Prod.	Copper (Cu, ore)	secondary	2.63e-04	7.21e-05				
Aluminum Prod.Bentonite (in ground)secondary2.47e-046.77e-05Styrene-butadiene Copolymer Prod.Bauxitesecondary2.10e-045.76e-05Polycarbonate ProductionBauxite (Al2O3, ore)secondary1.55e-044.25e-05Steel Prod., cold-rolled, semi-finishedPyrite (FeS2, ore)secondary1.49e-044.09e-05PET Resin ProductionCoal, lignite (in ground)secondary1.21e-043.32e-05Natural Gas Prod.Bauxite (Al2O3, ore)secondary6.28e-051.72e-05Styrene-butadiene Copolymer Prod.Sulfur (S, in ground)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary2.90e-057.97e-06Natural Gas Prod.Uranium (U, ore)secondary2.79e-057.67e-06PET Resin ProductionIron (Fe, ore)secondary2.79e-057.22e-06	Styrene-butadiene Copolymer Prod.	Iron (Fe, ore)	secondary	2.55e-04	7.00e-05				
Styrene-butadiene Copolymer Prod.Bauxitesecondary2.10e-045.76e-05Polycarbonate ProductionBauxite (Al2O3, ore)secondary1.55e-044.25e-05Steel Prod., cold-rolled, semi-finishedPyrite (FeS2, ore)secondary1.49e-044.09e-05PET Resin ProductionCoal, lignite (in ground)secondary1.21e-043.32e-05Natural Gas Prod.Bauxite (Al2O3, ore)secondary6.28e-051.72e-05Styrene-butadiene Copolymer Prod.Sulfur (S, in ground)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary2.90e-057.97e-06Natural Gas Prod.Uranium (U, ore)secondary2.79e-057.67e-06PET Resin ProductionIron (Fe, ore)secondary2.63e-057.22e-06	Aluminum Prod.	Bentonite (in ground)	secondary	2.47e-04	6.77e-05				
Polycarbonate ProductionBauxite (Al2O3, ore)secondary1.55e-044.25e-05Steel Prod., cold-rolled, semi-finishedPyrite (FeS2, ore)secondary1.49e-044.09e-05PET Resin ProductionCoal, lignite (in ground)secondary1.21e-043.32e-05Natural Gas Prod.Bauxite (Al2O3, ore)secondary6.28e-051.72e-05Styrene-butadiene Copolymer Prod.Sulfur (S, in ground)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary2.90e-057.97e-06Natural Gas Prod.Uranium (U, ore)secondary2.79e-057.67e-06PET Resin ProductionIron (Fe, ore)secondary2.63e-057.22e-06	Styrene-butadiene Copolymer Prod.	Bauxite	secondary	2.10e-04	5.76e-05				
Steel Prod., cold-rolled, semi-finishedPyrite (FeS2, ore)secondary1.49e-044.09e-05PET Resin ProductionCoal, lignite (in ground)secondary1.21e-043.32e-05Natural Gas Prod.Bauxite (Al2O3, ore)secondary6.28e-051.72e-05Styrene-butadiene Copolymer Prod.Sulfur (S, in ground)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary2.90e-057.97e-06Natural Gas Prod.Uranium (U, ore)secondary2.79e-057.67e-06PET Resin ProductionIron (Fe, ore)secondary2.63e-057.22e-06	Polycarbonate Production	Bauxite (Al2O3, ore)	secondary	1.55e-04	4.25e-05				
PET Resin ProductionCoal, lignite (in ground)secondary1.21e-043.32e-05Natural Gas Prod.Bauxite (Al2O3, ore)secondary6.28e-051.72e-05Styrene-butadiene Copolymer Prod.Sulfur (S, in ground)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary2.90e-057.97e-06Natural Gas Prod.Uranium (U, ore)secondary2.79e-057.67e-06PET Resin ProductionIron (Fe, ore)secondary2.63e-057.22e-06	Steel Prod., cold-rolled, semi-finished	Pyrite (FeS2, ore)	secondary	1.49e-04	4.09e-05				
Natural Gas Prod.Bauxite (Al2O3, ore)secondary6.28e-051.72e-05Styrene-butadiene Copolymer Prod.Sulfur (S, in ground)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary2.90e-057.97e-06Natural Gas Prod.Uranium (U, ore)secondary2.79e-057.67e-06PET Resin ProductionIron (Fe, ore)secondary2.63e-057.22e-06	PET Resin Production	Coal, lignite (in ground)	secondary	1.21e-04	3.32e-05				
Styrene-butadiene Copolymer Prod.Sulfur (S, in ground)secondary3.56e-059.78e-06PET Resin ProductionBauxite (Al2O3, ore)secondary2.90e-057.97e-06Natural Gas Prod.Uranium (U, ore)secondary2.79e-057.67e-06PET Resin ProductionIron (Fe, ore)secondary2.63e-057.22e-06	Natural Gas Prod.	Bauxite (Al2O3, ore)	secondary	6.28e-05	1.72e-05				
PET Resin ProductionBauxite (Al2O3, ore)secondary2.90e-057.97e-06Natural Gas Prod.Uranium (U, ore)secondary2.79e-057.67e-06PET Resin ProductionIron (Fe, ore)secondary2.63e-057.22e-06	Styrene-butadiene Copolymer Prod.	Sulfur (S, in ground)	secondary	3.56e-05	9.78e-06				
Natural Gas Prod.Uranium (U, ore)secondary2.79e-057.67e-06PET Resin ProductionIron (Fe, ore)secondary2.63e-057.22e-06	PET Resin Production	Bauxite (Al2O3, ore)	secondary	2.90e-05	7.97e-06				
PET Resin Production Iron (Fe, ore) secondary 2.63e-05 7.22e-06	Natural Gas Prod.	Uranium (U, ore)	secondary	2.79e-05	7.67e-06				
	PET Resin Production	Iron (Fe, ore)	secondary	2.63e-05	7.22e-06				

Table M-4. LCD LCIA Results for the Nonrenewable Resource Use Impact Category
Process Group	Material	LCI Data Type	Nonrenewable Resource Use (kg)	% of Total
Styrene-butadiene Copolymer Prod.	Bentonite (in ground)	secondary	2.53e-05	6.95e-06
Aluminum Prod.	Lead (Pb, ore)	secondary	2.35e-05	6.44e-06
PMMA Sheet Prod.	Potassium chloride (KCl, as K2O, in ground)	secondary	2.34e-05	6.42e-06
Steel Prod., cold-rolled, semi-finished	Uranium (U, ore)	secondary	2.09e-05	5.74e-06
Polycarbonate Production	Uranium (U, ore)	secondary	1.75e-05	4.79e-06
Aluminum Prod.	Uranium (U, ore)	secondary	1.07e-05	2.92e-06
Steel Prod., cold-rolled, semi-finished	Bentonite (in ground)	secondary	8.78e-06	2.41e-06
Steel Prod., cold-rolled, semi-finished	Iron sulfate (FeSO4, ore)	secondary	7.99e-06	2.19e-06
Polycarbonate Production	Calcium sulfate (CaSO4, ore)	secondary	7.22e-06	1.98e-06
Steel Prod., cold-rolled, semi-finished	Bauxite (Al2O3, ore)	secondary	5.79e-06	1.59e-06
PMMA Sheet Prod.	Fluorspar (CaF2, ore)	secondary	5.37e-06	1.47e-06
Polycarbonate Production	Dolomite (in ground)	secondary	5.16e-06	1.42e-06
Polycarbonate Production	Olivine ore	secondary	4.13e-06	1.13e-06
Styrene-butadiene Copolymer Prod.	Potassium chloride (KCl, as K2O, in ground)	secondary	3.80e-06	1.04e-06
Polycarbonate Production	Potassium chloride (KCl, as K2O, in ground)	secondary	3.09e-06	8.49e-07
PMMA Sheet Prod.	Olivine ore	secondary	3.07e-06	8.42e-07
PET Resin Production	Sulfur (S, in ground)	secondary	2.81e-06	7.72e-07
Styrene-butadiene Copolymer Prod.	Calcium sulfate (CaSO4, ore)	secondary	2.53e-06	6.95e-07
PET Resin Production	Uranium (U, ore)	secondary	1.66e-06	4.56e-07
Styrene-butadiene Copolymer Prod.	Dolomite (in ground)	secondary	1.09e-06	2.98e-07
Polycarbonate Production	Lead (Pb, ore)	secondary	1.03e-06	2.83e-07
PMMA Sheet Prod.	Bentonite (in ground)	secondary	7.67e-07	2.10e-07
Styrene-butadiene Copolymer Prod.	Olivine ore	secondary	7.23e-07	1.98e-07
PET Resin Production	Bentonite (in ground)	secondary	3.63e-07	9.96e-08
PET Resin Production	Dolomite (in ground)	secondary	3.63e-07	9.96e-08
PET Resin Production	Olivine	secondary	2.72e-07	7.47e-08
PET Resin Production	Lead (Pb, ore)	secondary	1.82e-07	4.98e-08
Steel Prod., cold-rolled, semi-finished	Copper (Cu, ore)	secondary	9.10e-08	2.50e-08
Steel Prod., cold-rolled, semi-finished	Lead (Pb, ore)	secondary	2.84e-08	7.79e-09
Steel Prod., cold-rolled, semi-finished	Chromium ore	secondary	1.79e-08	4.91e-09
Steel Prod., cold-rolled, semi-finished	Manganese (Mn, ore)	secondary	1.04e-08	2.86e-09
Steel Prod., cold-rolled, semi-finished	Nickel (Ni, ore)	secondary	6.05e-09	1.66e-09
Steel Prod., cold-rolled, semi-finished	Zinc (Zn, ore)	secondary	6.61e-10	1.81e-10
Steel Prod., cold-rolled, semi-finished	Silver (Ag, ore)	secondary	4.51e-10	1.24e-10
Total Materials Processing			2.50e+02	6.86e+01
Manufacturing Life-cycle Stage				
LPG Production	Petroleum (in ground)	secondary	1.80e+01	4.93e+00
Japanese Electric Grid	Coal, average (in ground)	model/secondary	7.68e+00	2.11e+00
Natural Gas Prod.	Natural gas (in ground)	secondary	5.36e+00	1.47e+00
Japanese Electric Grid	Petroleum (in ground)	model/secondary	4.33e+00	1.19e+00
Japanese Electric Grid	Natural gas	model/secondary	4.20e+00	1.15e+00
LPG Production	Natural gas (in ground)	secondary	2.16e+00	5.93e-01
LPG Production	Coal, average (in ground)	secondary	6.44e-01	1.77e-01
US electric grid	Coal, average (in ground)	model/secondary	3.47e-01	9.53e-02
Fuel Oil #4 Prod.	Petroleum (in ground)	secondary	2.26e-01	6.20e-02
Fuel Oil #6 Prod.	Petroleum (in ground)	secondary	1.32e-01	3.63e-02
Fuel Oil #2 Prod.	Petroleum (in ground)	secondary	5.77e-02	1.58e-02
Natural Gas Prod.	Coal, average (in ground)	secondary	3.67e-02	1.01e-02
US electric grid	Natural gas	model/secondary	2.71e-02	7.43e-03
Fuel Oil #4 Prod.	Natural gas (in ground)	secondary	2.06e-02	5.64e-03
Natural Gas Prod.	Petroleum (in ground)	secondary	1.32e-02	3.63e-03

Table M-4. LCD LCIA Results for the Nonrenewable Resource Use Impact Category

Process Group	Material	LCI Data Type	Nonrenewable Resource Use (kg)	% of Total
Fuel Oil #6 Prod.	Natural gas (in ground)	secondary	9.80e-03	2.69e-03
US electric grid	Petroleum (in ground)	model/secondary	7.36e-03	2.02e-03
Fuel Oil #2 Prod.	Natural gas (in ground)	secondary	6.35e-03	1.74e-03
Fuel Oil #6 Prod.	Coal, average (in ground)	secondary	3.69e-03	1.01e-03
LPG Production	Bauxite (Al2O3, ore)	secondary	2.13e-03	5.85e-04
Fuel Oil #2 Prod.	Coal, average (in ground)	secondary	1.96e-03	5.39e-04
Japanese Electric Grid	Uranium, yellowcake	model/secondary	1.02e-03	2.80e-04
Panel components	Borax	primary	9.13e-05	2.51e-05
Fuel Oil #4 Prod.	Bauxite (Al2O3, ore)	secondary	1.56e-05	4.29e-06
LPG Production	Uranium (U, ore)	secondary	1.08e-05	2.97e-06
US electric grid	Uranium, yellowcake	model/secondary	9.39e-06	2.58e-06
Fuel Oil #2 Prod.	Bauxite (Al2O3, ore)	secondary	5.82e-06	1.60e-06
Fuel Oil #6 Prod.	Bauxite (Al2O3, ore)	secondary	5.12e-06	1.41e-06
Natural Gas Prod.	Bauxite (Al2O3, ore)	secondary	1.42e-06	3.89e-07
Natural Gas Prod.	Uranium (U, ore)	secondary	6.31e-07	1.73e-07
Fuel Oil #6 Prod.	Uranium (U, ore)	secondary	6.21e-08	1.70e-08
Fuel Oil #2 Prod.	Uranium (U, ore)	secondary	3.31e-08	9.07e-09
Total Manufacturing			4.33e+01	1.19e+01
Use, Maintenance and Repair Lit	fe-cycle Stage			
US electric grid	Coal, average (in ground)	model/secondary	6.69e+01	1.84e+01
US electric grid	Natural gas	model/secondary	5.22e+00	1.43e+00
US electric grid	Petroleum (in ground)	model/secondary	1.42e+00	3.89e-01
US electric grid	Uranium, yellowcake	model/secondary	1.81e-03	4.97e-04
Total Use, Maintenance and Rep	air		7.36e+01	2.02e+01
End-of-life Life-cycle Stage				
US electric grid	Coal, average (in ground)	model/secondary	1.27e-02	3.49e-03
LCD landfilling	Oil (in ground)	primary	8.06e-03	2.21e-03
LPG Production	Petroleum (in ground)	secondary	1.48e-03	4.05e-04
LCD landfilling	Natural gas (in ground)	primary	1.09e-03	2.98e-04
US electric grid	Natural gas	model/secondary	9.91e-04	2.72e-04
LCD landfilling	Coal, average (in ground)	primary	7.38e-04	2.02e-04
LCD landfilling	Iron (Fe, ore)	primary	3.96e-04	1.09e-04
LCD incineration	Iron (Fe, ore)	secondary	3.07e-04	8.43e-05
US electric grid	Petroleum (in ground)	model/secondary	2.69e-04	7.39e-05
LPG Production	Natural gas (in ground)	secondary	1.78e-04	4.87e-05
LPG Production	Coal, average (in ground)	secondary	5.28e-05	1.45e-05
US electric grid	Uranium, yellowcake	model/secondary	3.44e-07	9.43e-08
LCD landfilling	Bauxite (Al2O3, ore)	primary	2.12e-07	5.82e-08
LPG Production	Bauxite (Al2O3, ore)	secondary	1.75e-07	4.81e-08
LCD landfilling	Uranium (U, ore)	primary	1.00e-08	2.75e-09
LPG Production	Uranium (U, ore)	secondary	8.88e-10	2.44e-10
Natural Gas Prod.	Uranium (U, ore)	secondary	-1.32e-07	-3.63e-08
Natural Gas Prod.	Bauxite (Al2O3, ore)	secondary	-2.98e-07	-8.16e-08
LCD incineration	Bauxite (Al2O3, ore)	secondary	-4.37e-06	-1.20e-06
Fuel Oil #4 Prod.	Bauxite (Al2O3, ore)	secondary	-7.20e-05	-1.98e-05
Natural Gas Prod.	Petroleum (in ground)	secondary	-2.78e-03	-7.62e-04
Natural Gas Prod.	Coal, average (in ground)	secondary	-7.71e-03	-2.12e-03
LCD incineration	Natural gas	secondary	-5.85e-02	-1.60e-02
Fuel Oil #4 Prod.	Natural gas (in ground)	secondary	-9.48e-02	-2.60e-02
Fuel Oil #4 Prod.	Petroleum (in ground)	secondary	-1.04e+00	-2.86e-01

Table M-4. LCD LCIA Results for the Nonrenewable Resource Use Impact Category

Process Group	Material	LCI Data Type	Nonrenewable Resource Use (kg)	% of Total
Natural Gas Prod.	Natural gas (in ground)	secondary	-1.13e+00	-3.09e-01
Total End-of-Life			-2.30e+00	-6.32e-01
Total All Life-cycle Stages			3.64e+02	1.00e+02

Table M-4. LCD LCIA Results for the Nonrenewable Resource Use Impact Category

Process Group	Material	LCI Data Type	Energy Use (MJ)	% of Total
Materials Processing Life-cycle Stag	e			
Polycarbonate Production	Natural gas (in ground)	secondary	3.42e+01	1.64e-01
Aluminum Prod.	Coal, average (in ground)	secondary	2.20e+01	1.05e-01
Steel Prod., cold-rolled, semi-finished	Petroleum (in ground)	secondary	2.10e+01	1.01e-01
Steel Prod., cold-rolled, semi-finished	Natural gas (in ground)	secondary	2.04e+01	9.79e-02
Aluminum Prod.	Electricity	secondary	2.03e+01	9.76e-02
Aluminum Prod.	Petroleum (in ground)	secondary	1.99e+01	9.54e-02
Styrene-butadiene Copolymer Prod.	Natural gas (in ground)	secondary	1.91e+01	9.19e-02
Steel Prod., cold-rolled, semi-finished	Electricity	secondary	1.81e+01	8.71e-02
Invar	Petroleum (in ground)	secondary	1.59e+01	7.64e-02
Invar	Natural gas (in ground)	secondary	1.56e+01	7.50e-02
Polycarbonate Production	Electricity	secondary	1.39e+01	6.68e-02
Steel Prod., cold-rolled, semi-finished	Coal, average (in ground)	secondary	1.08e+01	5.19e-02
Aluminum Prod.	Natural gas (in ground)	secondary	9.51e+00	4.57e-02
Polycarbonate Production	Petroleum (in ground)	secondary	9.49e+00	4.56e-02
Polycarbonate Production	Coal, average (in ground)	secondary	8.73e+00	4.19e-02
ABS Production	Natural gas (in ground)	secondary	8.71e+00	4.18e-02
Invar	Electricity	secondary	8.06e+00	3.87e-02
Styrene-butadiene Copolymer Prod.	Petroleum (in ground)	secondary	8.04e+00	3.86e-02
Lead	Natural gas (in ground)	secondary	6.96e+00	3.34e-02
Steel Prod cold-rolled semi-finished	Uranium (U ore)	secondary	6 44e+00	3 09e-02
Lead	Coal average (in ground)	secondary	6 43e+00	3 08e-02
Steel Prod cold-rolled semi-finished	Coal lignite (in ground)	secondary	6.17e+00	2.96e-02
ABS Production	Petroleum (in ground)	secondary	5.91e+00	2.90e 02
Ferrite mfg	Natural gas (in ground)	secondary	5.51e+00	2.64e-02
Polycarbonate Production	Uranium (U_ore)	secondary	4 72e+00	2.01e 02
Ferrite mfg	Flectricity	secondary	4.72c+00	2.27C-02
Aluminum Prod	Uranium (IL ore)	secondary	4.32e+00	2.13e-02
Polycarbonate Production	Coal lignite (in ground)	secondary	4.520+00	1 75e 02
I and	Electricity	secondary	3.14e+00	1.750-02
ABS Production	Electricity	secondary	3.00e+00	1.51C-02
Polystyrene Prod high-impact	Natural gas (in ground)	secondary	2.81e+00	1.44c-02
Aluminum Prod	Coal lignite (in ground)	secondary	2.810+00	1.30e.02
Lead	Petroleum (in ground)	secondary	2.71C+00	1.300-02
Ecau Formite mfg	Coal average (in ground)	secondary	2.090+00	1.290-02
Ferrite mfg.	Detroloum (in ground)	secondary	2.396+00	1.15e-02
Ferrite mfg.	Coal lignite (in ground)	secondary	2.10e+00	0.122.02
Ferrite mig.	Uranium (IL and)	secondary	1.900-00	9.12e-03
Petrite IIIg.	Dramum (0, ore)	secondary	1.600+00	7.72- 02
Stymme hutediane Construct	Petroleum (in ground)	secondary	1.01e+00	7.72e-03
Styrene-butadiene Copolymer Prod.		secondary	1.39e+00	0.07e-03
	Uranium (U, ore)	secondary	9.1/e-01	4.40e-03
Polystyrene Prod., high-impact	Electricity	secondary	7.29e-01	3.50e-03
Total Materials Processing			3.66e+02	1.75e+00
Manufacturing Life-cycle Stage				
Glass/frit	Liquified petroleum gas ("propane")	primary	1.51e+04	7.23e+01
LPG Production	Natural gas (in ground)	secondary	2.01e+03	9.63e+00
LPG Production	Petroleum (in ground)	secondary	4.10e+02	1.97e+00
LPG Production	Coal, average (in ground)	secondary	3.00e+02	1.44e+00
CRT tube mfg.	Fuel oil #6	primary	1.51e+02	7.23e-01
Glass/frit	Natural gas	primary	5.41e+01	2.60e-01
Glass/frit	Fuel oil #2	primary	5.06e+01	2.43e-01

 Table M-5. CRT LCIA Results for the Energy Use Impact Category

Process Group	Material	LCI Data Type	Energy Use (MJ)	% of Total
Glass/frit	Elec-nonlink	primary	4.75e+01	2.28e-01
CRT tube mfg.	Natural gas	primary	3.73e+01	1.79e-01
LPG Production	Uranium (U, ore)	secondary	3.40e+01	1.63e-01
CRT tube mfg.	Elec-nonlink	primary	3.19e+01	1.53e-01
Glass/frit	Elec-nonlink	primary	2.65e+01	1.27e-01
PWB Mfg.	Natural gas	model/secondary	1.74e+01	8.34e-02
CRT tube mfg.	LNG	primary	1.72e+01	8.26e-02
CRT monitor assembly	Elec-nonlink	primary	1.33e+01	6.37e-02
Fuel Oil #6 Prod.	Natural gas (in ground)	secondary	1.28e+01	6.15e-02
PWB Mfg.	Elec-nonlink	model/secondary	1.00e+01	4.82e-02
Fuel Oil #2 Prod.	Natural gas (in ground)	secondary	6.06e+00	2.91e-02
Natural Gas Prod.	Natural gas (in ground)	secondary	5.83e+00	2.80e-02
Fuel Oil #6 Prod.	Petroleum (in ground)	secondary	5.78e+00	2.78e-02
Monitor/module	Fuel oil #4	primary	5.76e+00	2.77e-02
Fuel Oil #6 Prod.	Coal, average (in ground)	secondary	2.42e+00	1.16e-02
Fuel Oil #2 Prod.	Coal, average (in ground)	secondary	9.41e-01	4.51e-03
Fuel Oil #4 Prod.	Natural gas (in ground)	secondary	5.95e-01	2.85e-03
Fuel Oil #2 Prod.	Petroleum (in ground)	secondary	5.78e-01	2.77e-03
Natural Gas Prod.	Coal, average (in ground)	secondary	5.21e-01	2.50e-03
Natural Gas Prod.	Petroleum (in ground)	secondary	3.60e-01	1.73e-03
Fuel Oil #6 Prod.	Uranium (U, ore)	secondary	2.75e-01	1.32e-03
Fuel Oil #4 Prod.	Petroleum (in ground)	secondary	2.14e-01	1.03e-03
Fuel Oil #2 Prod.	Uranium (U, ore)	secondary	1.07e-01	5.13e-04
Natural Gas Prod.	Uranium (U, ore)	secondary	6.04e-02	2.90e-04
Total Manufacturing			1.83e+04	8.79e+01
Use, Maintenance and Repair Life-cy	vcle Stage			
CRT monitor use	Elec-nonlink	primary	2.29e+03	1.10e+01
Total Use, Maintenance and Repair			2.29e+03	1.10e+01
End-of-life Manufacturing Stage				
CRT landfilling	Fuel oil #4	primary	8.13e-01	3.90e-03
CRT landfilling	Natural gas	primary	7.71e-01	3.70e-03
CRT Recycling	Elec-nonlink	primary	2.29e-01	1.10e-03
CRT Recycling	Liquified petroleum gas ("propane")	primary	1.30e-01	6.24e-04
LPG Production	Natural gas (in ground)	secondary	1.73e-02	8.32e-05
LPG Production	Petroleum (in ground)	secondary	3.54e-03	1.70e-05
LPG Production	Coal, average (in ground)	secondary	2.59e-03	1.24e-05
LPG Production	Uranium (U, ore)	secondary	2.94e-04	1.41e-06
Natural Gas Prod.	Uranium (U, ore)	secondary	-3.03e-02	-1.45e-04
Natural Gas Prod.	Petroleum (in ground)	secondary	-1.81e-01	-8.67e-04
Natural Gas Prod.	Coal, average (in ground)	secondary	-2.61e-01	-1.25e-03
Fuel Oil #4 Prod.	Petroleum (in ground)	secondary	-2.30e+00	-1.10e-02
Natural Gas Prod.	Natural gas (in ground)	secondary	-2.92e+00	-1.40e-02
Fuel Oil #4 Prod.	Natural gas (in ground)	secondary	-6.39e+00	-3.07e-02
CRT Incineration	Natural gas	secondary	-5.88e+01	-2.82e-01
CRT Incineration	Fuel oil #4	secondary	-5.88e+01	-2.82e-01
Total End-of-life			-1.28e+02	-6.13e-01
Total All Life-cycle Stages			2.08e+04	1.00e+02

Table M-5. CRT LCIA Result	ts for the	Energy	Use Im	pact (Category
------------------------------------	------------	--------	--------	--------	----------

Process Group	Material	LCI Data Type	Energy Use (MJ)	% of Total
Materials processing Life-cycle Stag	e			
Natural Gas Prod.	Natural gas (in ground)	secondary	4.07e+02	1.43e+01
Natural Gas Prod.	Coal, average (in ground)	secondary	3.64e+01	1.28e+00
Natural Gas Prod.	Petroleum (in ground)	secondary	2.52e+01	8.86e-01
Polycarbonate Production	Natural gas (in ground)	secondary	1.91e+01	6.72e-01
PMMA Sheet Prod.	Petroleum (in ground)	secondary	1.18e+01	4.17e-01
PMMA Sheet Prod.	Natural gas (in ground)	secondary	1.16e+01	4.09e-01
Steel Prod., cold-rolled, semi-finished	Petroleum (in ground)	secondary	1.03e+01	3.63e-01
Steel Prod., cold-rolled, semi-finished	Natural gas (in ground)	secondary	9.99e+00	3.52e-01
PMMA Sheet Prod.	Electricity	secondary	8.96e+00	3.16e-01
Steel Prod., cold-rolled, semi-finished	Electricity	secondary	8.89e+00	3.13e-01
Styrene-butadiene Copolymer Prod.	Natural gas (in ground)	secondary	8.37e+00	2.95e-01
Aluminum Prod.	Coal, average (in ground)	secondary	8.18e+00	2.88e-01
Polycarbonate Production	Electricity	secondary	7.77e+00	2.74e-01
Aluminum Prod.	Electricity	secondary	7.57e+00	2.67e-01
Aluminum Prod.	Petroleum (in ground)	secondary	7.40e+00	2.61e-01
Polycarbonate Production	Petroleum (in ground)	secondary	5.30e+00	1.87e-01
Steel Prod., cold-rolled, semi-finished	Coal, average (in ground)	secondary	5.30e+00	1.87e-01
Polycarbonate Production	Coal, average (in ground)	secondary	4.87e+00	1.72e-01
Natural Gas Prod.	Uranium (U, ore)	secondary	4.22e+00	1.48e-01
Aluminum Prod.	Natural gas (in ground)	secondary	3.54e+00	1.25e-01
Styrene-butadiene Copolymer Prod.	Petroleum (in ground)	secondary	3.51e+00	1.24e-01
Steel Prod., cold-rolled, semi-finished	Uranium (U, ore)	secondary	3.16e+00	1.11e-01
Steel Prod., cold-rolled, semi-finished	Coal, lignite (in ground)	secondary	3.02e+00	1.07e-01
Polycarbonate Production	Uranium (U, ore)	secondary	2.64e+00	9.28e-02
Polycarbonate Production	Coal, lignite (in ground)	secondary	2.04e+00	7.18e-02
Aluminum Prod.	Uranium (U, ore)	secondary	1.61e+00	5.66e-02
PET Resin Production	Petroleum (in ground)	secondary	1.57e+00	5.54e-02
Aluminum Prod.	Coal, lignite (in ground)	secondary	1.01e+00	3.55e-02
PET Resin Production	Coal, average (in ground)	secondary	8.58e-01	3.02e-02
PET Resin Production	Electricity	secondary	7.72e-01	2.72e-02
Styrene-butadiene Copolymer Prod.	Electricity	secondary	6.07e-01	2.14e-02
PET Resin Production	Natural gas (in ground)	secondary	3.16e-01	1.11e-02
PET Resin Production	Uranium (U, ore)	secondary	2.51e-01	8.82e-03
PET Resin Production	Coal, lignite (in ground)	secondary	1.79e-03	6.32e-05
Total Materials Processing			6 33e+02	2.23e+01
Manufacturing Life-cycle Stage			0.000	2.200.01
I CD glass mfg	Liquified petroleum gas ("propage")	nrimory	6.08e+02	$2.46_{2}+0.1$
LCD glass ling. Monitor/module	Elec nonlink	primary	2.54e+02	2.40e+01 8.93e+00
Monitor/module	LNG	primary	1.65e+02	5.81e+00
L DG Production	Natural cas (in ground)	pi lillai y	0.622+01	3.81e+00
Denal components	Flag poplink	secondary	9.05C+01	1.442+00
Monitor/module	Netural gas	primary	4.08C+01 2.772+01	1.440+00
Monitor/module	Liquified petroleum gas ("propage")	primary	2.50e+01	8.81e.01
I BG Production	Petroleum (in ground)	secondary	2.30C+01	6.03e.01
LPG Production	Coal average (in ground)	secondary	1.970+01	5.07e.01
Monitor/module	Kerosene	primary	1.44C+01	4.600.01
Natural Gas Prod	Natural gas (in ground)	primary secondary	0.10+00	3.240.01
Monitor/module	Fuel oil #4	primary	9.190+00 8.88e+00	3 13e-01
PWB Mfr	Natural gas	model/secondam	7.67e±00	2 70= 01
i wb wiig.	ivatural gas	model/secondary	7.07E+00	2.700-01

 Table M-6. LCD LCIA Results for the Energy Use Impact Category

Process Group	Material	LCI Data Type	Energy Use (MJ)	% of Total
Panel components	Kerosene	primary	7.36e+00	2.59e-01
Panel components	Elec-nonlink	primary	5.13e+00	1.81e-01
Panel components	Fuel oil #6	primary	5.13e+00	1.81e-01
Monitor/module	Elec-nonlink	primary	4.92e+00	1.73e-01
PWB Mfg.	Elec-nonlink	model/secondary	4.43e+00	1.56e-01
Panel components	Natural gas	primary	3.85e+00	1.35e-01
Backlight	Elec-nonlink	primary	2.62e+00	9.23e-02
LCD glass mfg.	Natural gas	primary	2.50e+00	8.82e-02
LCD glass mfg.	Fuel oil #2	primary	2.34e+00	8.26e-02
LCD glass mfg.	Elec-nonlink	primary	2.20e+00	7.75e-02
LPG Production	Uranium (U, ore)	secondary	1.63e+00	5.75e-02
Backlight	Elec-nonlink	primary	1.24e+00	4.36e-02
Fuel Oil #4 Prod.	Natural gas (in ground)	secondary	9.16e-01	3.23e-02
Natural Gas Prod.	Coal, average (in ground)	secondary	8.21e-01	2.89e-02
Backlight	Elec-nonlink	primary	6.01e-01	2.12e-02
Natural Gas Prod.	Petroleum (in ground)	secondary	5.68e-01	2.00e-02
Panel components	Elec-nonlink	primary	5.04e-01	1.77e-02
Fuel Oil #6 Prod.	Natural gas (in ground)	secondary	4.36e-01	1.54e-02
Fuel Oil #4 Prod.	Petroleum (in ground)	secondary	3.30e-01	1.16e-02
Fuel Oil #2 Prod.	Natural gas (in ground)	secondary	2.83e-01	9.96e-03
Fuel Oil #6 Prod.	Petroleum (in ground)	secondary	1.97e-01	6.93e-03
Panel components	Steam (100 psig)	primary	1.04e-01	3.66e-03
Natural Gas Prod.	Uranium (U, ore)	secondary	9.51e-02	3.35e-03
Fuel Oil #6 Prod.	Coal, average (in ground)	secondary	8.24e-02	2.90e-03
Fuel Oil #2 Prod.	Coal, average (in ground)	secondary	4.39e-02	1.55e-03
Fuel Oil #2 Prod.	Petroleum (in ground)	secondary	2.70e-02	9.50e-04
Panel components	Fuel oil #2	primary	1.78e-02	6.25e-04
Fuel Oil #6 Prod.	Uranium (U, ore)	secondary	9.37e-03	3.30e-04
Fuel Oil #2 Prod.	Uranium (U, ore)	secondary	4.99e-03	1.76e-04
Backlight	LNG	primary	2.14e-04	7.53e-06
Backlight	Natural gas	primary	1.85e-04	6.53e-06
Panel components	Natural gas	primary	5.24e-06	1.85e-07
Total Manufacturing			1.44e+03	5.06e+01
Use. Maintenance and Repair Life-cy	vcle Stage			
LCD monitor use	Elec-nonlink	primary	8 53e+02	3.00e+01
Total use Maintanance and Banair		printary	9.530102	3.000+01
For the second s			8.55e+02	5.00e+01
End-of-life Life-cycle Stage				
LCD landfilling	Fuel oil #4	primary	1.96e-01	6.90e-03
LCD landfilling	Natural gas	primary	1.86e-01	6.54e-03
LCD recycling	Elec-nonlink	primary	1.62e-01	5.70e-03
LCD recycling	Liquified petroleum gas ("propane")	primary	5.93e-02	2.09e-03
LPG Production	Natural gas (in ground)	secondary	7.91e-03	2.78e-04
LPG Production	Petroleum (in ground)	secondary	1.61e-03	5.69e-05
LPG Production	Coal, average (in ground)	secondary	1.18e-03	4.16e-05
LPG Production	Uranium (U, ore)	secondary	1.34e-04	4.72e-06
Natural Gas Prod.	Uranium (U, ore)	secondary	-2.00e-02	-7.04e-04
Natural Gas Prod.	Petroleum (in ground)	secondary	-1.19e-01	-4.20e-03
Natural Gas Prod.	Coal, average (in ground)	secondary	-1.72e-01	-6.07e-03
Fuel Oil #4 Prod.	Petroleum (in ground)	secondary	-1.52e+00	-5.35e-02
Natural Gas Prod.	Natural gas (in ground)	secondary	-1.93e+00	-6.80e-02
Fuel Oil #4 Prod.	Natural gas (in ground)	secondary	-4.22e+00	-1.49e-01

 Table M-6. LCD LCIA Results for the Energy Use Impact Category

Process Group	Material	LCI Data Type	Energy Use (MJ)	% of Total
LCD incineration	Natural gas	secondary	-3.85e+01	-1.36e+00
LCD incineration	Fuel oil #4	secondary	-3.85e+01	-1.36e+00
Total End-of-life			-8.44e+01	-2.97e+00
Total All Life-cycle Stages			2.84e+03	1.00e+02

Table M-6. LCD LCIA Results for the Energy Use Impact Category

Process Group	Material	LCI Data Type	Solid Waste Landfill Use (m3)	% of Total
Materials Processing Life-cycle Stage				
Steel Prod., cold-rolled, semi-finished	Unspecified solid waste	secondary	1.09e-02	6.56e+00
Steel Prod., cold-rolled, semi-finished	Mining waste	secondary	6.41e-04	3.85e-01
Invar	Mining waste	secondary	1.82e-04	1.09e-01
Ferrite mfg.	Mining waste	secondary	1.77e-04	1.06e-01
Steel Prod., cold-rolled, semi-finished	Mineral waste	secondary	6.34e-05	3.81e-02
Polycarbonate Production	Mineral waste	secondary	5.41e-05	3.25e-02
Steel Prod., cold-rolled, semi-finished	Mixed industrial (waste)	secondary	4.50e-05	2.70e-02
Invar	Unspecified solid waste	secondary	3.53e-05	2.12e-02
Ferrite mfg.	Unspecified solid waste	secondary	3.45e-05	2.07e-02
Steel Prod., cold-rolled, semi-finished	Slag and ash	secondary	1.83e-05	1.10e-02
Invar	Mineral waste	secondary	1.82e-05	1.09e-02
Polycarbonate Production	Slag and ash	secondary	1.80e-05	1.08e-02
Ferrite mfg.	Mineral waste	secondary	1.78e-05	1.07e-02
ABS Production	Mineral waste	secondary	1.27e-05	7.64e-03
ABS Production	Unspecified solid waste	secondary	1.09e-05	6.57e-03
Polycarbonate Production	Mixed industrial (waste)	secondary	9.93e-06	5.96e-03
Invar	Slag and ash	secondary	5.24e-06	3.15e-03
Ferrite mfg.	Slag and ash	secondary	5.12e-06	3.07e-03
Styrene-butadiene Copolymer Prod.	Mineral waste	secondary	4.52e-06	2.72e-03
	Non toxic chemical waste			
ABS Production	(unspecified)	secondary	3.21e-06	1.93e-03
ABS Production	Slag and ash	secondary	2.68e-06	1.61e-03
ABS Production	Mixed industrial (waste)	secondary	2.22e-06	1.34e-03
Steel Prod., cold-rolled, semi-finished	Non toxic chemical waste (unspecified)	secondary	2.02e-06	1.21e-03
	Non toxic chemical waste			
Polycarbonate Production	(unspecified)	secondary	1.79e-06	1.08e-03
Polystyrene Prod., high-impact	Mineral waste	secondary	1.59e-06	9.56e-04
Styrona hutadiana Canalymar Brad	Non toxic chemical waste	saaandami	1 250 06	8 002 04
Inver	(unspecified) Mixed industrial (wasta)	secondary	1.160.06	6.060.04
nivai Forrito mfa	Mixed industrial (waste)	secondary	1.100-00	6.802.04
Styrene butadiene Conolymer Prod	Slag and ash	secondary	1.130-00	6.54e.04
Styrene-butadiene Copolymer Prod.	Mixed inductrial (waste)	secondary	1.090-00	6.030.04
Styrene-butadiene Copolymer Frod.	Non toxia chamical waste	secondary	1.000-00	0.036-04
Polystyrene Prod., high-impact	(unspecified)	secondary	6.74e-07	4.05e-04
Polystyrene Prod., high-impact	Mixed industrial (waste)	secondary	4.16e-07	2.50e-04
Aluminum Prod.	Calcium	secondary	3.72e-07	2.23e-04
Polystyrene Prod., high-impact	Slag and ash	secondary	3.66e-07	2.20e-04
Styrene-butadiene Copolymer Prod.	Unspecified solid waste	secondary	2.62e-07	1.57e-04
Invar	Calcium	secondary	1.79e-07	1.07e-04
Polycarbonate Production	Unspecified solid waste	secondary	1.09e-07	6.55e-05
Lead	Calcium	secondary	9.31e-08	5.59e-05
Steel Prod., cold-rolled, semi-finished	Sewage sludge (unspecified)	secondary	6.64e-08	3.99e-05
Aluminum Prod.	Carbon	secondary	5.48e-08	3.29e-05
Steel Prod., cold-rolled, semi-finished	Waste oil	secondary	5.37e-08	3.23e-05
Steel Prod., cold-rolled, semi-finished	Non mineral waste (inert)	secondary	4.46e-08	2.68e-05
Ferrite mfg.	Calcium	secondary	3.41e-08	2.05e-05
Steel Prod., cold-rolled. semi-finished	Calcium	secondary	3.02e-08	1.81e-05
Invar	Carbon	secondary	2.62e-08	1.57e-05

Process Group	Material	LCI Data Type	Solid Waste Landfill Use (m3)	% of Total
Aluminum Prod.	Waste oil	secondary	2.25e-08	1.35e-05
Polystyrene Prod., high-impact	Unspecified solid waste	secondary	2.02e-08	1.21e-05
Invar	Non toxic chemical waste (unspecified)	secondary	1.54e-08	9.25e-06
Ferrite mfg.	Non toxic chemical waste (unspecified)	secondary	1.50e-08	9.03e-06
Lead	Carbon	secondary	1.36e-08	8.19e-06
Invar	Non mineral waste (inert)	secondary	1.08e-08	6.46e-06
Ferrite mfg.	Non mineral waste (inert)	secondary	1.05e-08	6.30e-06
Invar	Waste oil	secondary	8.51e-09	5.11e-06
Ferrite mfg.	Carbon	secondary	4.90e-09	2.94e-06
Steel Prod., cold-rolled, semi-finished	Carbon	secondary	4.32e-09	2.59e-06
Lead	Waste oil	secondary	4.30e-09	2.58e-06
Ferrite mfg.	Waste oil	secondary	2.72e-10	1.63e-07
Total Materials Processing			1.23e-02	7.39e+00
Manufacturing Life-cycle Stage				
LPG Production	Slag and ash	secondary	3.46e-02	2.08e+01
LPG Production	Unspecified waste	secondary	7.44e-03	4.47e+00
Glass/frit	Waste water treatment (WWT) sludge	primary	7.28e-03	4.37e+00
CRT tube mfg.	Waste water treatment (WWT) sludge	primary	2.37e-03	1.42e+00
LPG Production	Mixed industrial (waste)	secondary	1.24e-03	7.47e-01
US electric grid	Coal waste	model/secondary	1.01e-03	6.09e-01
Japanese Electric Grid	Coal waste	model/secondary	8.10e-04	4.86e-01
US electric grid	Dust/sludge	model/secondary	3.14e-04	1.89e-01
Fuel Oil #6 Prod.	Slag and ash	secondary	2.79e-04	1.68e-01
US electric grid	Fly/bottom ash	model/secondary	2.54e-04	1.52e-01
Japanese Electric Grid	Dust/sludge	model/secondary	2.51e-04	1.50e-01
Japanese Electric Grid	Fly/bottom ash	model/secondary	2.02e-04	1.22e-01
Fuel Oil #2 Prod.	Slag and ash	secondary	1.09e-04	6.52e-02
Natural Gas Prod.	Slag and ash	secondary	6.01e-05	3.61e-02
Fuel Oil #6 Prod.	Unspecified waste	secondary	6.00e-05	3.60e-02
Fuel Oil #2 Prod.	Unspecified waste	secondary	2.33e-05	1.40e-02
Natural Gas Prod.	Unspecified waste	secondary	1.29e-05	7.76e-03
Fuel Oil #4 Prod.	Slag and ash	secondary	1.16e-05	6.96e+00
Glass/frit	Potassium Carbonate	primary	1.08e-05	6.48e-03
CRT tube mfg.	Wood, average	primary	9.89e-06	5.94e-03
Glass/frit	Sodium Carbonate	primary	9.74e-06	5.85e-03
Glass/frit	CRT glass, faceplate	primary	9.73e-06	5.84e-03
Glass/frit	Unspecified sludge	primary	7.69e-06	4.62e-03
CRT tube mfg.	Sludge (phosphor)	primary	4.31e-06	2.59e-03
Fuel Oil #6 Prod.	Mixed industrial (waste)	secondary	4.23e-06	2.54e-03
Fuel Oil #2 Prod.	Mixed industrial (waste)	secondary	3.49e-06	2.10e-03
Fuel Oil #4 Prod.	Unspecified waste	secondary	2.49e-06	1.49e-03
CRT tube mfg.	Sludge (aquadag)	primary	2.22e-06	1.34e-03
LPG Production	Non toxic chemical waste (unspecified)	secondary	1.35e-06	8.13e-04
LPG Production	Mineral waste	secondary	1.01e-06	6.07e-04
Glass/frit	Waste refractory	primary	9.52e-07	5.72e-04
Glass/frit	blasting media	primary	8.16e-07	4.90e-04
Glass/frit	Oily rags & filter media	primary	7.25e-07	4.36e-04

	courts for the Sond Wast	e Landini Ose i	impact Catego	ı y
Process Group	Material	LCI Data Type	Solid Waste Landfill Use (m3)	% of Total
Glass/frit	Plating process sludge	primary	2.98e-07	1.79e-04
Fuel Oil #4 Prod.	Mixed industrial (waste)	secondary	2.84e-07	1.71e-04
Glass/frit	acid absorbent	primary	1.02e-07	6.11e-05
Natural Gas Prod.	Mixed industrial (waste)	secondary	2.51e-08	1.51e-05
Fuel Oil #6 Prod	Non toxic chemical waste	secondary	4 58e-09	2 75e-06
	Non toxic chemical waste	secondary	1.500 05	2.750 00
Fuel Oil #2 Prod.	(unspecified)	secondary	3.80e-09	2.28e-06
Fuel Oil #6 Prod.	Mineral waste	secondary	3.42e-09	2.05e-06
Fuel Oil #2 Prod.	Mineral waste	secondary	2.84e-09	1.70e-06
Fuel Oil #4 Prod	Non toxic chemical waste	sacandam	2.00- 10	1 850 07
Fuel Oil #4 Prod.	(unspecified)	secondary	3.09e-10	1.856-07
Fuel Oli #4 Prod.	Mineral waste	secondary	2.508-10	1.386-07
Natural Gas Prod.	(unspecified)	secondary	2.74e-11	1.64e-08
Natural Gas Prod	Mineral waste	secondary	2.04e-11	1.23e-08
Total Manufacturing		secondary	5.640.02	2 200 101
Use Meintenenes and Densin Life syste	Store		5.046-02	3.396+01
Use, Maintenance and Repair Life-cycle	Stage	1.1/ 1	(25 . 02	2.02 + 01
US electric grid		model/secondary	6.35e-02	3.82e+01
		model/secondary	1.97e-02	1.18e+01
	Fly/bottom ash	model/secondary	1.59e-02	9.54e+00
Total Use, Maintenance and Repair			9.91e-02	5.95e+01
End-of-life Life-cycle Stage				
CRT landfilling	EOL CRT Monitor, landfilled	primary	8.31e-03	4.99e+00
US electric grid	Coal waste	model/secondary	6.36e-06	3.82e-03
CRT landfilling	Unspecified solid waste	primary	4.34e-06	2.61e-03
CRT Incineration	Mixed industrial (waste)	secondary	2.41e-06	1.45e-03
US electric grid	Dust/sludge	model/secondary	1.97e-06	1.18e-03
US electric grid	Fly/bottom ash	model/secondary	1.59e-06	9.54e-04
LPG Production	Slag and ash	secondary	2.99e-07	1.79e-04
LPG Production	Unspecified waste	secondary	6.42e-08	3.85e-05
LPG Production	Mixed industrial (waste)	secondary	1.07e-08	6.44e-06
L PG Production	Non toxic chemical waste	secondary	1 17e 11	7.01e.09
L PG Production	Mineral waste	secondary	8 72e-12	5 24e-09
Natural Gas Prod	Mineral waste	secondary	-1.02e-11	-6 15e-09
	Non toxic chemical waste	secondary	1.020 11	0.150 05
Natural Gas Prod.	(unspecified)	secondary	-1.37e-11	-8.24e-09
CRT Incineration	Mineral waste	secondary	-1.59e-10	-9.55e-08
CRT Incineration	Non toxic chemical waste (unspecified)	secondary	-2.13e-10	-1.28e-07
Fuel Oil #4 Prod.	Mineral waste	secondary	-2.48e-09	-1.49e-06
E 101//4 B 1	Non toxic chemical waste		2.22 00	1.00.00
CDT In sinemation	(unspectified)	secondary	-3.32e-09	-1.99e-06
CKT Incineration	Mining waste	secondary	-4.23e-09	-2.54e-06
Fuel Oil #4 Drod	Mixed industrial (Waste)	secondary	-1.26e-08	-7.57e-06
Fuci Oli #4 Prod.	Unerpaified waste	secondary	-3.05e-06	-1.836-03
Fuel Oil #4 Drod	Unspecified waste	secondary	-0.486-00	-3.896-03
Fuci Oli #4 Flou.	Slog and ash	secondary	-2.0/e-05	-1.01e-02
Fuel Oil #4 Prod	Slag and ash	secondary	-5.01e-05	-1.816-02
1 uci ()11 #4 F10u.	Stag allu asli	secondary	-1.256-04	-7.466-02

Process Group	Material	LCI Data Type	Solid Waste Landfill Use (m3)	% of Total
CRT Incineration	Unspecified solid waste	secondary	-1.76e-03	-1.06e+00
CRT Incineration	Slag and ash	secondary	-7.70e-03	-4.62e+00
Total End-of-life			-1.32e-03	-7.95e-01
Total All Life-cycle Stages	Total		1.67e-01	1.00e+02

Process Group	Material	LCI Data Type	Solid Waste landfill Use (m3)	% of Total
Materials Processing Life-cycle St	age			
Steel Prod., cold-rolled, semi-finished	Unspecified solid waste	secondary	5.36e-03	9.87e+00
Natural Gas Prod.	Slag and ash	secondary	4.20e-03	7.74e+00
Natural Gas Prod.	Unspecified waste	secondary	9.03e-04	1.66e+00
Steel Prod., cold-rolled, semi-finished	Mining waste	secondary	3.14e-04	5.79e-01
Steel Prod., cold-rolled, semi-finished	Mineral waste	secondary	3.11e-05	5.73e-02
Polycarbonate Production	Mineral waste	secondary	3.02e-05	5.56e-02
PMMA Sheet Prod.	Mixed industrial (waste)	secondary	2.45e-05	4.51e-02
Steel Prod., cold-rolled, semi-finished	Mixed industrial (waste)	secondary	2.21e-05	4.06e-02
PMMA Sheet Prod.	Mineral waste	secondary	1.99e-05	3.67e-02
Polycarbonate Production	Slag and ash	secondary	1.00e-05	1.85e-02
PMMA Sheet Prod.	Non toxic chemical waste (unspecified)	secondary	9.41e-06	1.73e-02
Steel Prod., cold-rolled, semi-finished	Slag and ash	secondary	8.99e-06	1.66e-02
PMMA Sheet Prod.	Slag and ash	secondary	6.26e-06	1.15e-02
Polycarbonate Production	Mixed industrial (waste)	secondary	5.55e-06	1.02e-02
PET Resin Production	Mineral waste	secondary	2.87e-06	5.29e-03
Styrene-butadiene Copolymer Prod.	Mineral waste	secondary	1.98e-06	3.64e-03
Natural Gas Prod.	Mixed industrial (waste)	secondary	1.76e-06	3.24e-03
PET Resin Production	Slag and ash	secondary	1.24e-06	2.29e-03
Polycarbonate Production	Non toxic chemical waste (unspecified)	secondary	1.00e-06	1.84e-03
Steel Prod., cold-rolled, semi-finished	Non toxic chemical waste (unspecified)	secondary	9.88e-07	1.82e-03
PET Resin Production	Non toxic chemical waste (unspecified)	secondary	7.69e-07	1.42e-03
Styrene-butadiene Copolymer Prod.	Non toxic chemical waste (unspecified)	secondary	5.89e-07	1.09e-03
Styrene-butadiene Copolymer Prod.	Slag and ash	secondary	4.76e-07	8.77e-04
Styrene-butadiene Copolymer Prod.	Mixed industrial (waste)	secondary	4.39e-07	8.08e-04
PET Resin Production	Unspecified solid waste	secondary	3.65e-07	6.73e-04
Aluminum Prod.	Calcium	secondary	1.38e-07	2.55e-04
Styrene-butadiene Copolymer Prod.	Unspecified solid waste	secondary	1.15e-07	2.11e-04
PMMA Sheet Prod.	Unspecified solid waste	secondary	7.91e-08	1.46e-04
Polycarbonate Production	Unspecified solid waste	secondary	6.10e-08	1.12e-04
Steel Prod., cold-rolled, semi-finished	Sewage sludge (unspecified)	secondary	3.25e-08	6.00e-05
Steel Prod., cold-rolled, semi-finished	Waste oil	secondary	2.63e-08	4.85e-05
Steel Prod., cold-rolled, semi-finished	Non mineral waste (inert)	secondary	2.19e-08	4.03e-05
Aluminum Prod.	Carbon	secondary	2.04e-08	3.76e-05
Steel Prod., cold-rolled, semi-finished	Calcium	secondary	1.48e-08	2.73e-05
Aluminum Prod.	Waste oil	secondary	8.38e-09	1.54e-05
Steel Prod., cold-rolled, semi-finished	Carbon	secondary	2.12e-09	3.90e-06
Natural Gas Prod.	Non toxic chemical waste (unspecified)	secondary	1.91e-09	3.52e-06
Natural Gas Prod.	Mineral waste	secondary	1.43e-09	2.63e-06
Total Materials Processing			1.10e-02	2.02e+01
Manufacturing Life-cycle Stage				
Monitor/module	Waste water treatment (WWT) sludge	primary	3.06e-03	5.65e+00
Japanese Electric Grid	Coal waste	model/secondary	2.72e-03	5.02e+00
LPG Production	Slag and ash	secondary	1.66e-03	3.06e+00
Japanese Electric Grid	Dust/sludge	model/secondary	8.43e-04	1.55e+00
Japanese Electric Grid	Fly/bottom ash	model/secondary	6.81e-04	1.25e+00
LPG Production	Unspecified waste	secondary	3.57e-04	6.57e-01
LCD glass mfg.	Waste water treatment (WWT) sludge	primary	3.39e-04	6.25e-01
Monitor/module	Waste plastics from LCD monitor	primary	2.89e-04	5.33e-01
Monitor/module	Waste acid (containing F and detergents)	primary	1.80e-04	3.31e-01

 Table M-8. LCD LCIA Results for the Solid Waste Landfill Use Impact Category

Process Group	Material	LCI Data Type	Solid Waste landfill Use (m3)	% of Total
US electric grid	Coal waste	model/secondary	1.23e-04	2.27e-01
Natural Gas Prod.	Slag and ash	secondary	9.47e-05	1.75e-01
Monitor/module	Waste LCD glass	primary	8.23e-05	1.52e-01
LPG Production	Mixed industrial (waste)	secondary	5.97e-05	1.10e-01
US electric grid	Dust/sludge	model/secondary	3.81e-05	7.03e-02
US electric grid	Fly/bottom ash	model/secondary	3.08e-05	5.67e-02
Panel components	Waste LCD glass	primary	2.30e-05	4.23e-02
Natural Gas Prod.	Unspecified waste	secondary	2.04e-05	3.75e-02
Fuel Oil #4 Prod.	Slag and ash	secondary	1.78e-05	3.29e-02
Fuel Oil #6 Prod.	Slag and ash	secondary	9.52e-06	1.75e-02
Monitor/module	Printed wiring board (PWB)	primary	9.38e-06	1.73e-02
Fuel Oil #2 Prod.	Slag and ash	secondary	5.07e-06	9.34e-03
Fuel Oil #4 Prod.	Unspecified waste	secondary	3.83e-06	7.07e-03
Monitor/module	LCD panel waste	primary	3.79e-06	6.98e-03
Panel components	Used silica gel	primary	2.14e-06	3.94e-03
Fuel Oil #6 Prod.	Unspecified waste	secondary	2.04e-06	3.77e-03
Backlight	Waste backlight light guide (PMMA)	primary	1.28e-06	2.35e-03
Fuel Oil #2 Prod.	Unspecified waste	secondary	1.09e-06	2.01e-03
LCD glass mfg.	Potassium Carbonate	primary	4.99e-07	9.19e-04
LCD glass mfg.	Sodium Carbonate	primary	4.51e-07	8.31e-04
Fuel Oil #4 Prod.	Mixed industrial (waste)	secondary	4.38e-07	8.07e-04
LCD glass mfg.	Unspecified sludge	primary	3.56e-07	6.56e-04
LCD glass mfg.	Cinders from LCD glass mfg	primary	3.19e-07	5.87e-04
Fuel Oil #2 Prod.	Mixed industrial (waste)	secondary	1.63e-07	3.00e-04
Fuel Oil #6 Prod.	Mixed industrial (waste)	secondary	1.44e-07	2.65e-04
LCD glass mfg.	LCD glass EP dust	primary	9.94e-08	1.83e-04
LPG Production	Non toxic chemical waste (unspecified)	secondary	6.49e-08	1.20e-04
LPG Production	Mineral waste	secondary	4.85e-08	8.94e-05
LCD glass mfg.	Waste refractory	primary	4.41e-08	8.13e-05
Natural Gas Prod.	Mixed industrial (waste)	secondary	3.96e-08	7.30e-05
LCD glass mfg.	blasting media	primary	3.78e-08	6.97e-05
LCD glass mfg.	Waste LCD glass	primary	3.48e-08	6.41e-05
LCD glass mfg.	Oily rags & filter media	primary	3.36e-08	6.19e-05
LCD glass mfg.	Sludge from LCD glass mfg	primary	2.99e-08	5.51e-05
LCD glass mfg.	Plating process sludge	primary	1.38e-08	2.54e-05
Backlight	Waste backlight casing (PC)	primary	5.41e-09	9.97e-06
LCD glass mfg.	acid absorbent	primary	4.71e-09	8.68e-06
Fuel Oil #4 Prod.	Non toxic chemical waste (unspecified)	secondary	4.76e-10	8.76e-07
Fuel Oil #4 Prod.	Mineral waste	secondary	3.55e-10	6.54e-07
Fuel Oil #2 Prod.	Non toxic chemical waste (unspecified)	secondary	1.77e-10	3.27e-07
Fuel Oil #6 Prod.	Non toxic chemical waste (unspecified)	secondary	1.56e-10	2.87e-07
Fuel Oil #2 Prod.	Mineral waste	secondary	1.32e-10	2.44e-07
Fuel Oil #6 Prod.	Mineral waste	secondary	1.16e-10	2.14e-07
Natural Gas Prod.	Non toxic chemical waste (unspecified)	secondary	4.31e-11	7.95e-08
Natural Gas Prod.	Mineral waste	secondary	3.22e-11	5.93e-08
Backlight	Broken CCFL	primary	1.98e-11	3.65e-08
Total Manufacturing			1.07e-02	1.97e+01
Use, Maintenance and Repair Lif	e-cycle Stage			
US electric grid	Coal waste	model/secondary	2.37e-02	4.37e+01
US electric grid	Dust/sludge	model/secondary	7.35e-03	1.35e+01
-				

Table M-8. LCD LCIA Results for the Solid Waste Landfill Use Impact Category

Process Group	Material	LCI Data Type	Solid Waste landfill Use (m3)	% of Total
US electric grid	Fly/bottom ash	model/secondary	5.93e-03	1.09e+01
Total Use, Maintenance and Re	pair		3.70e-02	6.82e+01
End-of-life Life-cycle Stage				
LCD landfilling	EOL LCD Monitor, landfilled	primary	1.90e-03	3.50e+00
US electric grid	Coal waste	model/secondary	4.50e-06	8.29e-03
US electric grid	Dust/sludge	model/secondary	1.39e-06	2.57e-03
US electric grid	Fly/bottom ash	model/secondary	1.12e-06	2.07e-03
LCD landfilling	Unspecified solid waste	primary	1.05e-06	1.93e-03
LCD incineration	Mixed industrial (waste)	secondary	3.28e-07	6.04e-04
LPG Production	Slag and ash	secondary	1.36e-07	2.51e-04
LPG Production	Unspecified waste	secondary	2.93e-08	5.40e-05
LPG Production	Mixed industrial (waste)	secondary	4.90e-09	9.02e-06
LPG Production	Non toxic chemical waste (unspecified)	secondary	5.33e-12	9.82e-09
LPG Production	Mineral waste	secondary	3.98e-12	7.34e-09
Natural Gas Prod.	Mineral waste	secondary	-6.76e-12	-1.25e-08
Natural Gas Prod.	Non toxic chemical waste (unspecified)	secondary	-9.06e-12	-1.67e-08
LCD incineration	Mineral waste	secondary	-1.03e-10	-1.90e-07
LCD incineration	Non toxic chemical waste (unspecified)	secondary	-1.38e-10	-2.54e-07
Fuel Oil #4 Prod.	Mineral waste	secondary	-1.64e-09	-3.02e-06
Fuel Oil #4 Prod.	Non toxic chemical waste (unspecified)	secondary	-2.19e-09	-4.04e-06
LCD incineration	Mining waste	secondary	-2.74e-09	-5.06e-06
Natural Gas Prod.	Mixed industrial (waste)	secondary	-8.32e-09	-1.53e-05
Fuel Oil #4 Prod.	Mixed industrial (waste)	secondary	-2.02e-06	-3.72e-03
Natural Gas Prod.	Unspecified waste	secondary	-4.28e-06	-7.88e-03
Fuel Oil #4 Prod.	Unspecified waste	secondary	-1.77e-05	-3.26e-02
Natural Gas Prod.	Slag and ash	secondary	-1.99e-05	-3.67e-02
Fuel Oil #4 Prod.	Slag and ash	secondary	-8.22e-05	-1.52e-01
LCD incineration	Unspecified solid waste	secondary	-1.14e-03	-2.10e+00
LCD incineration	Slag and ash	secondary	-4.99e-03	-9.19e+00
Total End-of-life			-4.35e-03	-8.01e+00
Total All Life-cycle Stages			5.43e-02	1.00e+02

Process Group	Material	LCI Data Type	Hazardous Waste Landfill Use (m3)	% of Total
Materials Processing Life-cycle S	Stage			
Invar	General Hazardous Waste	secondary	3.92e-05	2.33e-01
Ferrite mfg.	General Hazardous Waste	secondary	3.82e-05	2.28e-01
Polycarbonate Production	General Hazardous Waste	secondary	2.06e-05	1.23e-01
ABS Production	General Hazardous Waste	secondary	9.45e-06	5.62e-02
Styrene-butadiene Copolymer Prod.	Hazardous waste	secondary	8.58e-07	5.11e-03
Steel Prod., cold-rolled, semi-finished	General Hazardous Waste	secondary	4.35e-07	2.59e-03
Polystyrene Prod., high-impact	General Hazardous Waste	secondary	2.70e-07	1.60e-03
Total Materials Processing			1.09e-04	6.49e-01
Manufacturing Life-cycle Stage				
LPG Production	Hazardous waste	secondary	1.36e-03	8.11e+00
Glass/frit	Hydrofluoric acid	primary	1.34e-05	7.98e-02
Glass/frit	cinders from CRT glass mfg (70% PbO)	primary	6.88e-06	4.09e-02
CRT tube mfg.	Unspecified sludge	primary	5.22e-06	3.11e-02
Fuel Oil #6 Prod.	Hazardous waste	secondary	4.61e-06	2.74e-02
Fuel Oil #2 Prod.	Hazardous waste	secondary	3.82e-06	2.27e-02
CRT tube mfg.	Frit	primary	3.04e-06	1.81e-02
CRT tube mfg.	Carbon and filmer waste	primary	2.30e-06	1.37e-02
Glass/frit	CRT glass faceplate EP dust (Pb) (D008 waste)	primary	2.15e-06	1.28e-02
CRT tube mfg.	Slurry scrap (chromium-based)	primary	1.60e-06	9.52e-03
Glass/frit	Hazardous sludge (Pb) (D008)	primary	1.38e-06	8.21e-03
CRT tube mfg.	Slag and ash	primary	1.30e-06	7.74e-03
CRT tube mfg.	Waste water treatment (WWT) filters	primary	7.59e-07	4.52e-03
Glass/frit	sludge from CRT glass mfg (1% PbO)	primary	6.45e-07	3.84e-03
Glass/frit	Broken CRT glass	primary	6.22e-07	3.70e-03
Glass/frit	Barium debris (D008 waste)	primary	4.58e-07	2.72e-03
Fuel Oil #4 Prod.	Hazardous waste	secondary	3.11e-07	1.85e-03
Glass/frit	Waste finishing sludge (Pb) (D008 waste)	primary	2.32e-07	1.38e-03
Glass/frit	Waste Batch (Ba, Pb) (D008 waste)	primary	1.22e-07	7.28e-04
CRT tube mfg.	Lead sulfate cake	primary	3.03e-08	1.80e-04
Natural Gas Prod.	Hazardous waste	secondary	2./5e-08	1.64e-04
Glass/frit	Lead debris (D008 waste)	primary	1.85e-08	1.10e-04
Total Manufacturing	Lead contaminated grit (Doos waste)	primary	2.996-09	1.786-03
	ferra alla Citara a		1.41e-03	8.40e+00
Use, Maintenance and Repair Li	ie-cycle Stage			
Total Use, Maintenance and Rep	air		0.00e+00	0.00e+00
End-of-life Life-cycle Stage				
CRT landfilling	EOL CRT Monitor, landfilled	primary	1.53e-02	9.10e+01
LPG Production	Hazardous waste	secondary	1.18e-08	7.00e-05
Natural Gas Prod.	Hazardous waste	secondary	-1.38e-08	-8.22e-05
CRT Incineration	General Hazardous Waste	secondary	-2.14e-07	-1.28e-03
Fuel Oil #4 Prod.	Hazardous waste	secondary	-3.34e-06	-1.99e-02
Total End-of-life			1.53e-02	9.10e+01
Total All Life-cycle Stages			1.68e-02	1.00e+02

Process Group	Material	LCI Data Type	Hazardous Waste Landfill Use (m3)	% of Total
Materials Processing Life-cycle Stag	e			
Polycarbonate Production	General Hazardous Waste	secondary	1.15e-05	3.18e-01
Natural Gas Prod.	Hazardous waste	secondary	1.92e-06	5.33e-02
PMMA Sheet Prod.	General Hazardous Waste	secondary	9.41e-07	2.60e-02
Styrene-butadiene Copolymer Prod.	Hazardous waste	secondary	3.75e-07	1.04e-02
Steel Prod., cold-rolled, semi-finished	General Hazardous Waste	secondary	2.13e-07	5.90e-03
PET Resin Production	Unspecified hazardous waste	secondary	3.24e-08	8.96e-04
Total Materials Processing			1.50e-05	4.15e-01
Manufacturing Life-cycle Stage				
LPG Production	Hazardous waste	secondary	6.54e-05	1.81e+00
Monitor/module	Acetic acid	primary	3.18e-05	8.80e-01
Monitor/module	Phosphoric acid	primary	8.55e-06	2.37e-01
LCD glass mfg.	Hydrofluoric acid	primary	6.21e-07	1.72e-02
Fuel Oil #4 Prod.	Hazardous waste	secondary	4.79e-07	1.33e-02
Monitor/module	Nitric acid	primary	2.48e-07	6.87e-03
Fuel Oil #2 Prod.	Hazardous waste	secondary	1.78e-07	4.94e-03
Fuel Oil #6 Prod.	Hazardous waste	secondary	1.57e-07	4.34e-03
Natural Gas Prod.	Hazardous waste	secondary	4.34e-08	1.20e-03
LCD glass mfg.	Barium debris (D008 waste)	primary	2.12e-08	5.87e-04
LCD glass mfg.	Waste Batch (Ba, Pb) (D008 waste)	primary	5.67e-09	1.57e-04
Backlight	Waste glass, with mercury	primary	7.73e-15	2.14e-10
Total Manufacturing			1.07e-04	2.98e+00
Use, Maintenance and Repair Life-cy	ycle Stage			
Total Use, Maintenance and Repair			0.00e+00	0.00e+00
End-of-life Life-cycle Stage				
LCD landfilling	EOL LCD Monitor, landfilled	primary	3.49e-03	9.67e+01
LPG Production	Hazardous waste	secondary	5.37e-09	1.49e-04
Natural Gas Prod.	Hazardous waste	secondary	-9.12e-09	-2.52e-04
LCD incineration	General Hazardous Waste	secondary	-1.39e-07	-3.84e-03
Fuel Oil #4 Prod.	Hazardous waste	secondary	-2.21e-06	-6.11e-02
Total End-of-life			3.49e-03	9.66e+01
Total All Life-cycle Stages			3.61e-03	1.00e+02

Table M-10. LCD LCIA Results for the Hazardous Waste Landfill Use Impact Category

Process Group	Material	LCI Data Type	Radioactive Waste Landfill Use (m3)	% of Total
Materials Processing Life-cycle St	age			
Steel Prod., cold-rolled, semi-finished	Low-level radioactive waste	secondary	1.64e-05	9.03e+00
Invar	Low-level radioactive waste	secondary	4.72e-06	2.60e+00
Ferrite mfg.	Low-level radioactive waste	secondary	4.61e-06	2.54e+00
Steel Prod., cold-rolled, semi-finished	Radioactive waste (unspecified)	secondary	7.36e-07	4.06e-01
Steel Prod., cold-rolled, semi-finished	Highly radioactive waste (Class C)	secondary	3.46e-07	1.91e-01
Invar	Radioactive waste (unspecified)	secondary	2.20e-07	1.22e-01
Ferrite mfg.	Radioactive waste (unspecified)	secondary	2.15e-07	1.19e-01
Invar	Highly radioactive waste (Class C)	secondary	9.80e-08	5.41e-02
Ferrite mfg.	Highly radioactive waste (Class C)	secondary	9.57e-08	5.28e-02
Total Materials Processing			2.74e-05	1.51e+01
Manufacturing Life-cycle Stage				
Japanese Electric Grid	Low-level radioactive waste	model/secondary	6.89e-06	3.80e+00
Japanese Electric Grid	Uranium, depleted	model/secondary	2.07e-06	1.14e+00
US electric grid	Low-level radioactive waste	model/secondary	1.75e-06	9.66e-01
US electric grid	Uranium, depleted	model/secondary	5.25e-07	2.90e-01
Total Manufacturing			1.12e-05	6.19e+00
Use, Maintenance and Repair Life	e-cycle Stage			
US electric grid	Low-level radioactive waste	model/secondary	1.10e-04	6.05e+01
US electric grid	Uranium, depleted	model/secondary	3.29e-05	1.82e+01
Total use, Maintenance and Repair	ir		1.43e-04	7.87e+01
End-of-life Life-cycle Stage				
US electric grid	Low-level radioactive waste	model/secondary	1.10e-08	6.06e-03
US electric grid	Uranium, depleted	model/secondary	3.29e-09	1.82e-03
Total End-of-life			1.43e-08	7.87e-03
Total All Life-cycle Stages			1.81e-04	1.00e+02

Table M-12.	LCD LCIA	Results for the	Radioactive	Waste]	Landfill	Use Im	pact Categ	gory
-------------	----------	------------------------	-------------	---------	----------	--------	------------	------

Process Group	Material	LCI Data Type	Radioactive Waste Landfill Use (m3)	% of Total
Materials Processing Life-cycle Stag	ge			
Steel Prod., cold-rolled, semi-finished	Low-level radioactive waste	secondary	8.02e-06	8.70e+00
Steel Prod., cold-rolled, semi-finished	Radioactive waste (unspecified)	secondary	3.61e-07	3.91e-01
Steel Prod., cold-rolled, semi-finished	Highly radioactive waste (Class C)	secondary	1.70e-07	1.84e-01
Total Materials Processing			8.55e-06	9.28e+00
Manufacturing Life-cycle Stage				
Japanese Electric Grid	Low-level radioactive waste	model/secondary	2.32e-05	2.51e+01
Japanese Electric Grid	Uranium, depleted	model/secondary	6.95e-06	7.54e+00
US electric grid	Low-level radioactive waste	model/secondary	2.12e-07	2.30e-01
US electric grid	Uranium, depleted	model/secondary	6.37e-08	6.91e-02
Total Manufacturing			3.04e-05	3.30e+01
Use, Maintenance and Repair Life-o	cycle Stage			
US electric grid	Low-level radioactive waste	model/secondary	4.10e-05	4.44e+01
US electric grid	Uranium, depleted	model/secondary	1.23e-05	1.33e+01
Total use, Maintenance and Repair			5.32e-05	5.77e+01
End-of-life Life-cycle Stage				
US electric grid	Low-level radioactive waste	model/secondary	7.77e-09	8.43e-03
US electric grid	Uranium, depleted	model/secondary	2.33e-09	2.53e-03
Total End-of-life			1.01e-08	1.10e-02
Total All Life-cycle Stages			9.22e-05	1.00e+02

Process Group	Material	LCI Data Type	Global Warming Potential (kg-Co2 Equivalents)	% of Total
Materials Processing Life-cycle Sta	ge			
Steel Prod., cold-rolled, semi-finished	Carbon dioxide	secondary	1.34e+01	1.93e+00
Polycarbonate Production	Carbon dioxide	secondary	4.62e+00	6.64e-01
Aluminum Prod.	Carbon dioxide	secondary	3.59e+00	5.16e-01
Invar	Carbon dioxide	secondary	2.26e+00	3.25e-01
Styrene-butadiene Copolymer Prod.	Carbon dioxide	secondary	1.65e+00	2.38e-01
ABS Production	Carbon dioxide	secondary	1.31e+00	1.89e-01
Lead	Carbon dioxide	secondary	1.07e+00	1.54e-01
Aluminum Prod.	Perfluoromethane	secondary	8.42e-01	1.21e-01
Ferrite mfg.	Carbon dioxide	secondary	8.32e-01	1.20e-01
Polycarbonate Production	Methane	secondary	4.27e-01	6.13e-02
Polystyrene Prod., high-impact	Carbon dioxide	secondary	4.23e-01	6.09e-02
Steel Prod., cold-rolled, semi-finished	Nitrous oxide	secondary	2.41e-01	3.46e-02
Steel Prod., cold-rolled, semi-finished	Methane	secondary	2.29e-01	3.29e-02
Aluminum Prod.	Methane	secondary	1.69e-01	2.43e-02
Styrene-butadiene Copolymer Prod.	Methane	secondary	1.44e-01	2.07e-02
Aluminum Prod.	Perfluoroethane	secondary	1.32e-01	1.90e-02
Invar	Methane	secondary	1.21e-01	1.74e-02
ABS Production	Methane	secondary	1.07e-01	1.53e-02
Ferrite mfg.	Methane	secondary	5.99e-02	8.62e-03
Lead	Methane	secondary	5.29e-02	7.61e-03
Aluminum Prod	Nitrous oxide	secondary	4 28e-02	6 16e-03
Polystyrene Prod., high-impact	Methane	secondary	3.49e-02	5.02e-03
Lead	Nitrous oxide	secondary	6 29e-03	9 05e-04
Ferrite mfg	Nitrous oxide	secondary	3 29e-03	4 73e-04
Polycarbonate Production	Nitrous oxide	secondary	1 43e-04	2.06e-03
ABS Production	Nitrous oxide	secondary	6 56e-05	9 44e-06
Styrene-butadiene Copolymer Prod	Nitrous oxide	secondary	6.41e-05	9.22e-06
Invar	Perfluoromethane	secondary	1.85e-05	2.67e-06
Ferrite mfg	Perfluoromethane	secondary	1.81e-05	2.60e-06
Invar	Nitrous oxide	secondary	1.50e-05	2.16e-06
Invar	HEC-125	secondary	6 43e-06	9 25e-07
Ferrite mfg	HFC-125	secondary	6 28e-06	9.03e-07
Steel Prod cold-rolled semi-finished	Perfluoromethane	secondary	2 92e-07	4 20e-08
Invar	Perfluoroethane	secondary	4 92e-08	7.08e-09
Total Matarials Processing	1 erridor o enhaño	secondary	2 19-101	4 57-100
Manufacturing Life evelo Stage			3.180+01	4.5/e+00
	0 1 1 1	1	1.51.00	2.17 +01
LPG Production	Carbon dioxide	secondary	1.51e+02	2.1/e+01
LPG Production	Methane	secondary	1.76e+01	2.53e+00
Japanese Electric Grid	Carbon dioxide	model/secondary	1.54e+01	2.21e+00
US electric grid	Carbon dioxide	model/secondary	7.10e+00	1.02e+00
LPG Production	Nitrous oxide	secondary	5.03e+00	7.23e-01
Glass/frit	Carbon dioxide	primary	2.81e+00	4.04e-01
Natural Gas Prod.	Carbon dioxide	secondary	1.32e+00	1.90e-01
Natural Gas Prod.	Methane	secondary	1.05e+00	1.51e-01
Fuel Oil #6 Prod.	Carbon dioxide	secondary	8.51e-01	1.22e-01
Fuel Oil #2 Prod.	Carbon dioxide	secondary	4.48e-01	6.44e-02
US electric grid	Methane	model/secondary	2.16e-01	3.11e-02
Fuel Oil #6 Prod.	Methane	secondary	1.24e-01	1.78e-02
Fuel Oil #2 Prod.	Methane	secondary	5.40e-02	7.76e-03

Table M-13. CRT LCIA Results for the Global Warming Impact Category

Process Group	Material	LCI Data Type	Global Warming Potential (kg-Co2 Equivalents)	% of Total
Fuel Oil #6 Prod.	Nitrous oxide	secondary	4.50e-02	6.47e-03
Glass/frit	Carbon dioxide	primary	4.33e-02	6.23e-03
Fuel Oil #4 Prod.	Carbon dioxide	secondary	4.22e-02	6.06e-03
Japanese Electric Grid	Nitrous oxide	model/secondary	3.46e-02	4.98e-03
US electric grid	Nitrous oxide	model/secondary	1.68e-02	2.42e-03
Fuel Oil #2 Prod.	Nitrous oxide	secondary	1.61e-02	2.31e-03
Fuel Oil #4 Prod.	Methane	secondary	5.47e-03	7.87e-04
Natural Gas Prod.	Nitrous oxide	secondary	3.58e-03	5.15e-04
Fuel Oil #4 Prod.	Nitrous oxide	secondary	1.78e-03	2.56e-04
Japanese Electric Grid	Methane	model/secondary	1.71e-03	2.46e-04
Total Manufacturing			2.03e+02	2.92e+01
Use, Maintenance and Repair Life-cy	ycle Stage			
US electric grid	Carbon dioxide	model/secondary	4.45e+02	6.39e+01
US electric grid	Methane	model/secondary	1.35e+01	1.95e+00
US electric grid	Nitrous oxide	model/secondary	1.05e+00	1.51e-01
Total Use, Maintenance and Repair			4.59e+02	6.60e+01
End-of-life Life-cycle Stage				
CRT Incineration	Carbon dioxide	secondary	3.55e+00	5.11e-01
CRT landfilling	Carbon dioxide	primary	1.03e-01	1.49e-02
US electric grid	Carbon dioxide	model/secondary	4.45e-02	6.40e-03
US electric grid	Methane	model/secondary	1.35e-03	1.95e-04
LPG Production	Carbon dioxide	secondary	1.30e-03	1.87e-04
CRT landfilling	Methane	primary	1.14e-03	1.64e-04
LPG Production	Methane	secondary	1.52e-04	2.19e-05
US electric grid	Nitrous oxide	model/secondary	1.05e-04	1.52e-05
LPG Production	Nitrous oxide	secondary	4.34e-05	6.24e-06
Natural Gas Prod.	Nitrous oxide	secondary	-1.80e-03	-2.58e-04
Fuel Oil #4 Prod.	Nitrous oxide	secondary	-1.92e-02	-2.75e-03
Fuel Oil #4 Prod.	Methane	secondary	-5.88e-02	-8.46e-03
CRT Incineration	Methane	secondary	-3.18e-01	-4.57e-02
CRT Incineration	Nitrous oxide	secondary	-3.23e-01	-4.64e-02
Fuel Oil #4 Prod.	Carbon dioxide	secondary	-4.53e-01	-6.52e-02
Natural Gas Prod.	Methane	secondary	-5.28e-01	-7.60e-02
Natural Gas Prod.	Carbon dioxide	secondary	-6.63e-01	-9.54e-02
Total End-of-life			1.34e+00	1.93e-01
Total All Life-cycle Stages			6.95e+02	1.00e+02

Table M-13. CRT LCIA Results for the Global Warming Impact Category

Process Group	Material	LCI Data Type	Global Warming Potential (kg-Co2 Equivalents)	% of Total
Materials Processing Life-cycle Stage				
Natural Gas Prod.	Carbon dioxide	secondary	9.24e+01	1.56e+01
Natural Gas Prod.	Methane	secondary	7.36e+01	1.24e+01
Steel Prod., cold-rolled, semi-finished	Carbon dioxide	secondary	6.58e+00	1.11e+00
PMMA Sheet Prod.	Carbon dioxide	secondary	2.65e+00	4.46e-01
Polycarbonate Production	Carbon dioxide	secondary	2.58e+00	4.35e-01
Aluminum Prod.	Carbon dioxide	secondary	1.34e+00	2.25e-01
Styrene-butadiene Copolymer Prod.	Carbon dioxide	secondary	7.23e-01	1.22e-01
PET Resin Production	Carbon dioxide	secondary	3.90e-01	6.58e-02
Aluminum Prod.	Perfluoromethane	secondary	3.14e-01	5.29e-02
Natural Gas Prod.	Nitrous oxide	secondary	2.50e-01	4.22e-02
PMMA Sheet Prod.	Methane	secondary	2.42e-01	4.07e-02
Polycarbonate Production	Methane	secondary	2.38e-01	4.02e-02
Steel Prod., cold-rolled, semi-finished	Nitrous oxide	secondary	1.18e-01	1.99e-02
Steel Prod., cold-rolled, semi-finished	Methane	secondary	1.12e-01	1.89e-02
Styrene-butadiene Copolymer Prod.	Methane	secondary	6.30e-02	1.06e-02
Aluminum Prod.	Methane	secondary	6.29e-02	1.06e-02
Aluminum Prod.	Perfluoroethane	secondary	4.93e-02	8.31e-03
PET Resin Production	Methane	secondary	2.10e-02	3.53e-03
Aluminum Prod.	Nitrous oxide	secondary	1.59e-02	2.69e-03
Polycarbonate Production	Nitrous oxide	secondary	7.99e-05	1.35e-05
PMMA Sheet Prod.	Nitrous oxide	secondary	5.94e-05	1.00e-05
PET Resin Production	Nitrous oxide	secondary	2.81e-05	4.74e-06
Styrene-butadiene Copolymer Prod.	Nitrous oxide	secondary	2.80e-05	4.73e-06
Steel Prod., cold-rolled, semi-finished	Perfluoromethane	secondary	1.43e-07	2.41e-08
Total Materials Processing			1.82e+02	3.06e+01
Manufacturing Life-cycle Stage				
Monitor/module	Sulfur hexafluoride	primary	1.74e+02	2.94e+01
Japanese Electric Grid	Carbon dioxide	model/secondary	5.18e+01	8.72e+00
LPG Production	Carbon dioxide	secondary	7.24e+00	1.22e+00
Natural Gas Prod.	Carbon dioxide	secondary	2.09e+00	3.51e-01
Natural Gas Prod.	Methane	secondary	1.66e+00	2.80e-01
US electric grid	Carbon dioxide	model/secondary	8.61e-01	1.45e-01
LPG Production	Methane	secondary	8.45e-01	1.42e-01
LPG Production	Nitrous oxide	secondary	2.41e-01	4.07e-02
LCD glass mfg.	Carbon dioxide	primary	1.30e-01	2.19e-02
Japanese Electric Grid	Nitrous oxide	model/secondary	1.17e-01	1.96e-02
Fuel Oil #4 Prod.	Carbon dioxide	secondary	6.50e-02	1.10e-02
Fuel Oil #6 Prod.	Carbon dioxide	secondary	2.90e-02	4.88e-03
US electric grid	Methane	model/secondary	2.62e-02	4.42e-03
Fuel Oil #2 Prod.	Carbon dioxide	secondary	2.09e-02	3.52e-03
Fuel Oil #4 Prod.	Methane	secondary	8.43e-03	1.42e-03
Japanese Electric Grid	Methane	model/secondary	5.74e-03	9.68e-04
Natural Gas Prod.	Nitrous oxide	secondary	5.64e-03	9.51e-04
Panel components	Carbon dioxide	primary	4.82e-03	8.12e-04
Fuel Oil #6 Prod.	Methane	secondary	4.21e-03	7.10e-04
Fuel Oil #4 Prod.	Nitrous oxide	secondary	2.75e-03	4.63e-04
Fuel Oil #2 Prod.	Methane	secondary	2.52e-03	4.24e-04
Monitor/module	Carbon dioxide	primary	2.16e-03	3.64e-04

Table M-14. LCD LCIA Results for the Global Warming Impact Category

Process Group	Material	LCI Data Type	Global Warming Potential (kg-Co2 Equivalents)	% of Total
US electric grid	Nitrous oxide	model/secondary	2.04e-03	3.44e-04
Fuel Oil #6 Prod.	Nitrous oxide	secondary	1.53e-03	2.58e-04
Fuel Oil #2 Prod.	Nitrous oxide	secondary	7.50e-04	1.26e-04
Total Manufacturing			2.40e+02	4.04e+01
Use, Maintenance and Repair Life-cycl	e Stage			
US electric grid	Carbon dioxide	model/secondary	1.66e+02	2.80e+01
US electric grid	Methane	model/secondary	5.05e+00	8.52e-01
US electric grid	Nitrous oxide	model/secondary	3.93e-01	6.63e-02
Total Use, Maintenance and Repair			1.71e+02	2.89e+01
End-of-life Life-cycle Stage				
LCD incineration	Carbon dioxide	secondary	2.07e+00	3.48e-01
US electric grid	Carbon dioxide	model/secondary	3.15e-02	5.31e-03
LCD landfilling	Carbon dioxide	primary	2.49e-02	4.20e-03
US electric grid	Methane	model/secondary	9.59e-04	1.62e-04
LPG Production	Carbon dioxide	secondary	5.95e-04	1.00e-04
LCD landfilling	Methane	primary	2.75e-04	4.64e-05
US electric grid	Nitrous oxide	model/secondary	7.46e-05	1.26e-05
LPG Production	Methane	secondary	6.94e-05	1.17e-05
LPG Production	Nitrous oxide	secondary	1.98e-05	3.34e-06
Natural Gas Prod.	Nitrous oxide	secondary	-1.19e-03	-2.00e-04
Fuel Oil #4 Prod.	Nitrous oxide	secondary	-1.27e-02	-2.13e-03
Fuel Oil #4 Prod.	Methane	secondary	-3.89e-02	-6.55e-03
LCD incineration	Methane	secondary	-2.07e-01	-3.49e-02
LCD incineration	Nitrous oxide	secondary	-2.09e-01	-3.53e-02
Fuel Oil #4 Prod.	Carbon dioxide	secondary	-2.99e-01	-5.05e-02
Natural Gas Prod.	Methane	secondary	-3.49e-01	-5.88e-02
Natural Gas Prod.	Carbon dioxide	secondary	-4.38e-01	-7.38e-02
Total End-of-life			5.70e-01	9.60e-02
Total All Life-cycle Stages			5.93e+02	1.00e+02

Table M-14. LCD LCIA Results for the Global Warming Impact Category

Process Group	Materials	LCI Data Type	Ozone Depletion (kg CFC-11 Equivalents)	% of Total
Materials Processing Life-cycle St	tage			
ABS Production	HALON-1301	secondary	4.10e-06	2.00e+01
Aluminum Prod.	HALON-1301	secondary	2.89e-06	1.41e+01
Invar	HALON-1301	secondary	1.22e-06	5.93e+00
Lead	HALON-1301	secondary	5.42e-07	2.65e+00
Steel Prod., cold-rolled, semi-finished	HALON-1301	secondary	4.03e-07	1.97e+00
Ferrite mfg.	HALON-1301	secondary	1.52e-07	7.43e-01
Invar	Dichlorodifluoromethane	secondary	8.02e-10	3.92e-03
Invar	CFC-13	secondary	5.04e-10	2.46e-03
ABS Production	HCFC-22	secondary	1.46e-10	7.14e-04
Ferrite mfg.	Dichlorodifluoromethane	secondary	5.94e-17	2.90e-10
Ferrite mfg.	CFC-13	secondary	3.74e-17	1.83e-10
Invar	HCFC-22	secondary	3.68e-18	1.80e-11
Ferrite mfg.	HCFC-22	secondary	3.59e-18	1.75e-11
Total Materials Processing			9.30e-06	4.54e+01
Manufacturing Life-cycle Stage			,	
L DG Production	Promomethane	sacandam	7 400 07	2.662+00
LFG Floduction	Bromomethane	model/secondemy	1.600.07	7 820 01
Japanasa Electric Grid	Bromomethane	model/secondary	1.000-07	6 25 2 01
I DG Production	1 1 1 Trichloroothano	secondary	1.200-07	7.840.02
LPG Production	HALON 1201	secondary	8 120 00	2 060 02
Japanese Electric Grid	1 1 1 Trichloroothano	model/secondem/	7 760 00	3.702.02
Fuel Oil #6 Bred	Promomethane	socondary	6.050.00	2.050.02
US electric grid	1 1 1 Trichloroethane	model/secondary	3.672.09	1 79e 02
Eval Oil #2 Brod	Promomethane	socondary	2 350 00	1.796-02
Netural Gas Prod	Bromomethana	secondary	1 300 00	6.342.02
Fuel Oil #4 Prod	Bromomethane	secondary	2.51e.10	1 220 03
Fuel Oil #6 Prod	1 1 1 Trichloroothano	secondary	1.302.10	6.220-03
Fuel Oil #2 Prod	1,1,1-Trichloroothane	secondary	5.042.11	2.462.04
Natural Gas Prod	1,1,1-Trichloroethane	secondary	2 70e 11	1 36e 04
Fuel Oil #6 Prod	HALON 1201	secondary	2.750-11	1.300-04
Fuel Oil #2 Brod	HALON 1201	secondary	2.750-11	1.110.04
Fuel Oil #4 Prod	1 1 1 Trichloroethane	secondary	5 37e 12	2.620.05
Fuel Oil #4 Prod	HALON 1301	secondary	1.85-12	9.040.06
Natural Gas Prod	HALON 1301	secondary	1.650-12	9.04c-00
Tatal Manuela studie	TIALON-1301	secondary	1.04-15	5.010-07
I otal Manufacturing	1.0		1.08e-06	5.29e+00
Use, Maintenance and Repair Life	e-cycle Stage			
US electric grid	Bromomethane	model/secondary	1.00e-05	4.90e+01
US electric grid	1,1,1-Trichloroethane	model/secondary	2.30e-07	1.12e+00
Total Use, Maintenance and Repa	ir		1.03e-05	5.01e+01
End-of-life Life-cycle Stage				
CRT Incineration	Carbon tetrachloride	secondary	1.61e-09	7.88e-03
CRT landfilling	Carbon tetrachloride	primary	1.51e-09	7.35e-03
US electric grid	Bromomethane	model/secondary	1.00e-09	4.91e-03
US electric grid	1,1,1-Trichloroethane	model/secondary	2.30e-11	1.12e-04
LPG Production	Bromomethane	secondary	6.47e-12	3.16e-05
LPG Production	1,1,1-Trichloroethane	secondary	1.39e-13	6.77e-07
LPG Production	HALON-1301	secondary	7.01e-14	3.42e-07
Natural Gas Prod.	HALON-1301	secondary	-8.23e-14	-4.02e-07

Table M-15. CRT LCIA Results for the Stratospheric Ozone Depletion Impact Category

	in itesuits for the structs		Depiecion impuer ou	
Process Group	Materials	LCI Data Type	Ozone Depletion (kg CFC-11 Equivalents)	% of Total
CRT Incineration	HALON-1301	secondary	-1.28e-12	-6.24e-06
Natural Gas Prod.	1,1,1-Trichloroethane	secondary	-1.40e-11	-6.82e-05
Fuel Oil #4 Prod.	HALON-1301	secondary	-1.99e-11	-9.72e-05
Fuel Oil #4 Prod.	1,1,1-Trichloroethane	secondary	-5.77e-11	-2.82e-04
Natural Gas Prod.	Bromomethane	secondary	-6.52e-10	-3.18e-03
Fuel Oil #4 Prod.	Bromomethane	secondary	-2.69e-09	-1.32e-02
CRT Incineration	1,1,1-Trichloroethane	secondary	-3.57e-09	-1.74e-02
CRT Incineration	Bromomethane	secondary	-1.67e-07	-8.13e-01
Total End-of-life			-1.69e-07	-8.27e-01
Total All Life-cycle Stages			2.05e-05	1.00e+02

Table M-15. CRT LCIA Results for the Stratospheric Ozone Depletion Impact Category

Process Group	Material	LCI Data Type	Ozone Depletion (kg CFC-11 Equivalents)	% of Total
Materials Processing Life-cycle Stag	e			
Aluminum Prod.	HALON-1301	secondary	1.07e-06	7.83e+00
Steel Prod., cold-rolled, semi-finished	HALON-1301	secondary	1.98e-07	1.44e+00
Natural Gas Prod.	Bromomethane	secondary	9.08e-08	6.62e-01
Natural Gas Prod.	1,1,1-Trichloroethane	secondary	1.95e-09	1.42e-02
Natural Gas Prod.	HALON-1301	secondary	1.15e-11	8.36e-05
Total Materials Processing			1.36e-06	9.95e+00
Manufacturing Life-cycle Stage				
Panel components	HCFC-225cb	primary	4.62e-06	3.37e+01
Panel components	HCFC-225ca	primary	3.50e-06	2.55e+01
Japanese Electric Grid	Bromomethane	model/secondary	4.31e-07	3.14e+00
LPG Production	Bromomethane	secondary	3.60e-08	2.62e-01
Japanese Electric Grid	1,1,1-Trichloroethane	model/secondary	2.61e-08	1.90e-01
US electric grid	Bromomethane	model/secondary	1.94e-08	1.42e-01
Natural Gas Prod.	Bromomethane	secondary	2.05e-09	1.49e-02
LPG Production	1,1,1-Trichloroethane	secondary	7.71e-10	5.62e-03
US electric grid	1,1,1-Trichloroethane	model/secondary	4.45e-10	3.25e-03
LPG Production	HALON-1301	secondary	3.90e-10	2.84e-03
Fuel Oil #4 Prod.	Bromomethane	secondary	3.86e-10	2.82e-03
Fuel Oil #6 Prod.	Bromomethane	secondary	2.06e-10	1.50e-03
Fuel Oil #2 Prod.	Bromomethane	secondary	1.10e-10	8.00e-04
Natural Gas Prod.	1,1,1-Trichloroethane	secondary	4.39e-11	3.20e-04
Fuel Oil #4 Prod.	1,1,1-Trichloroethane	secondary	8.28e-12	6.03e-05
Fuel Oil #6 Prod.	1,1,1-Trichloroethane	secondary	4.41e-12	3.22e-05
Fuel Oil #4 Prod.	HALON-1301	secondary	2.85e-12	2.08e-05
Fuel Oil #2 Prod.	1,1,1-Trichloroethane	secondary	2.35e-12	1.71e-05
Fuel Oil #2 Prod.	HALON-1301	secondary	1.06e-12	7.75e-06
Fuel Oil #6 Prod.	HALON-1301	secondary	9.35e-13	6.82e-06
Natural Gas Prod.	HALON-1301	secondary	2.59e-13	1.89e-06
Total Manufacturing			8.63e-06	6.29e+01
Use, Maintenance and Repair Life-c	ycle Stage			
US electric grid	Bromomethane	model/secondary	3.75e-06	2.73e+01
US electric grid	1,1,1-Trichloroethane	model/secondary	8.59e-08	6.26e-01
Total use, Maintenance and Repair			3.83e-06	2.79e+01
End-of-life Life-cycle Stage				
US electric grid	Bromomethane	model/secondary	7.11e-10	5.18e-03
LCD landfilling	Carbon tetrachloride	primary	3.63e-10	2.64e-03
LCD incineration	Carbon tetrachloride	secondary	2.81e-10	2.05e-03
US electric grid	1,1,1-Trichloroethane	model/secondary	1.63e-11	1.19e-04
LPG Production	Bromomethane	secondary	2.95e-12	2.15e-05
LPG Production	1,1,1-Trichloroethane	secondary	6.33e-14	4.61e-07
LPG Production	HALON-1301	secondary	3.20e-14	2.33e-07
Natural Gas Prod.	HALON-1301	secondary	-5.44e-14	-3.96e-07
LCD incineration	HALON-1301	secondary	-8.28e-13	-6.03e-06
Natural Gas Prod.	1,1,1-Trichloroethane	secondary	-9.22e-12	-6.72e-05
Fuel Oil #4 Prod.	HALON-1301	secondary	-1.32e-11	-9.58e-05
Fuel Oil #4 Prod.	1,1,1-Trichloroethane	secondary	-3.81e-11	-2.78e-04
Natural Gas Prod.	Bromomethane	secondary	-4.30e-10	-3.14e-03
Fuel Oil #4 Prod.	Bromomethane	secondary	-1.78e-09	-1.30e-02

Table M-16. LCD LCIA Results for the Stratospheric Ozone Depletion Impact Category

Process Group	Material	LCI Data Type	Ozone Depletion (kg CFC-11 Equivalents)	% of Total
LCD incineration	1,1,1-Trichloroethane	secondary	-2.31e-09	-1.69e-02
LCD incineration	Bromomethane	secondary	-1.08e-07	-7.86e-01
Total End-of-life			-1.11e-07	-8.10e-01
Total All Life-cycle Stages			1.37e-05	1.00e+02

Table M-16. LCD LCIA Results for the Stratospheric Ozone Depletion Impact Category

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Materials Processing Life-cycle	Stage		, ,,	
Steel Prod., cold-rolled, semi-finished	Nonmethane hydrocarbons, remaining unspeciated	secondary	3.32e-02	1.94e+01
Steel Prod., cold-rolled, semi-finished	Hydrocarbons, remaining unspeciated	secondary	3.44e-03	2.01e+00
ABS Production	Nonmethane hydrocarbons, remaining unspeciated	secondary	2.14e-03	1.25e+00
Polycarbonate Production	Nonmethane hydrocarbons, remaining unspeciated	secondary	1.84e-03	1.07e+00
Aluminum Prod.	Nonmethane hydrocarbons, remaining unspeciated	secondary	1.47e-03	8.60e-01
ABS Production	Hydrocarbons, remaining unspeciated	secondary	7.08e-04	4.14e-01
Styrene-butadiene Copolymer Prod.	Hydrocarbons, remaining unspeciated	secondary	6.59e-04	3.85e-01
Invar	Nonmethane hydrocarbons, remaining unspeciated	secondary	5.46e-04	3.19e-01
Lead	Nonmethane hydrocarbons, remaining unspeciated	secondary	3.66e-04	2.14e-01
Invar	VOCs, remaining unspeciated	secondary	2.31e-04	1.35e-01
Polystyrene Prod., high-impact	Hydrocarbons, remaining unspeciated	secondary	2.29e-04	1.34e-01
Ferrite mfg.	VOCs, remaining unspeciated	secondary	2.25e-04	1.32e-01
Steel Prod., cold-rolled, semi-finished	Ethylene	secondary	2.07e-04	1.21e-01
Ferrite mfg.	Nonmethane hydrocarbons, remaining unspeciated	secondary	1.84e-04	1.08e-01
ABS Production	Aromatic hydrocarbons	secondary	1.45e-04	8.48e-02
Polycarbonate Production	Methane	secondary	1.42e-04	8.31e-02
Invar	Ethylene	secondary	1.01e-04	5.93e-02
Steel Prod., cold-rolled, semi-finished	Ethane	secondary	9.79e-05	5.72e-02
Ferrite mfg.	Ethylene	secondary	9.77e-05	5.71e-02
Steel Prod., cold-rolled, semi-finished	Methane	secondary	7.63e-05	4.46e-02
Polycarbonate Production	Aromatic hydrocarbons	secondary	6.25e-05	3.66e-02
Aluminum Prod.	Methane	secondary	5.63e-05	3.29e-02
Styrene-butadiene Copolymer Prod.	Aromatic hydrocarbons	secondary	5.16e-05	3.02e-02
Styrene-butadiene Copolymer Prod.	Methane	secondary	4.81e-05	2.81e-02
ABS Production	Ethane	secondary	4.72e-05	2.76e-02
Invar	Methane	secondary	4.03e-05	2.36e-02
Steel Prod., cold-rolled, semi-finished	n-Propane	secondary	3.83e-05	2.24e-02
ABS Production	Methane	secondary	3.56e-05	2.08e-02
Invar	Ethane	secondary	3.29e-05	1.92e-02
Ferrite mfg.	Ethane	secondary	3.21e-05	1.88e-02
Polycarbonate Production	Aldehydes	secondary	2.62e-05	1.53e-02
ABS Production	Ethylene	secondary	2.57e-05	1.50e-02
Steel Prod., cold-rolled, semi-finished	Propylene	secondary	2.49e-05	1.45e-02
Polystyrene Prod., high-impact	Aromatic hydrocarbons	secondary	2.30e-05	1.35e-02
Invar	Hydrocarbons, remaining unspeciated	secondary	2.21e-05	1.29e-02
Ferrite mfg.	Hydrocarbons, remaining unspeciated	secondary	2.16e-05	1.26e-02
Ferrite mfg.	Methane	secondary	2.00e-05	1.17e-02
Steel Prod., cold-rolled, semi-finished	Aldehydes	secondary	1.89e-05	1.10e-02
Lead	Methane	secondary	1.76e-05	1.03e-02
Lead	Ethylene	secondary	1.59e-05	9.31e-03
Aluminum Prod.	Xylene (mixed isomers)	secondary	1.41e-05	8.23e-03
Aluminum Prod.	Polycyclic aromatic hydrocarbons	secondary	1.38e-05	8.06e-03
Lead	Ethane	secondary	1.26e-05	7.35e-03
Invar	n-Propane	secondary	1.24e-05	7.26e-03
Ferrite mfg.	n-Propane	secondary	1.21e-05	7.09e-03
Polystyrene Prod., high-impact	Methane	secondary	1.16e-05	6.80e-03
Steel Prod., cold-rolled, semi-finished	Butane	secondary	9.05e-06	5.29e-03
Invar	Propylene	secondary	7.07e-06	4.14e-03

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Steel Prod., cold-rolled, semi-finished	Pentane	secondary	6.92e-06	4.04e-03
Ferrite mfg.	Propylene	secondary	6.90e-06	4.04e-03
ABS Production	Formaldehyde	secondary	5.89e-06	3.44e-03
Invar	Xylene (mixed isomers)	secondary	5.53e-06	3.24e-03
Steel Prod., cold-rolled, semi-finished	Benzene	secondary	5.26e-06	3.07e-03
ABS Production	Heptane	secondary	4.74e-06	2.77e-03
Aluminum Prod.	Toluene	secondary	4.67e-06	2.73e-03
Aluminum Prod.	Aldehydes	secondary	4.59e-06	2.69e-03
Steel Prod., cold-rolled, semi-finished	Toluene	secondary	4.37e-06	2.55e-03
Steel Prod., cold-rolled, semi-finished	Xylene (mixed isomers)	secondary	4.16e-06	2.43e-03
Steel Prod., cold-rolled, semi-finished	Acetylene	secondary	3.91e-06	2.29e-03
Invar	Butane	secondary	3.80e-06	2.22e-03
Ferrite mfg.	Butane	secondary	3.71e-06	2.17e-03
Invar	Pentane	secondary	3.66e-06	2.14e-03
Ferrite mfg.	Pentane	secondary	3.57e-06	2.09e-03
Invar	Benzene	secondary	2.59e-06	1.52e-03
Lead	Xylene (mixed isomers)	secondary	2.59e-06	1.51e-03
Steel Prod., cold-rolled, semi-finished	Hexane	secondary	1.84e-06	1.08e-03
Ferrite mfg.	Benzene	secondary	1.76e-06	1.03e-03
Invar	Toluene	secondary	1.60e-06	9.33e-04
Ferrite mfg.	Toluene	secondary	1.56e-06	9.11e-04
Steel Prod., cold-rolled, semi-finished	Formaldehyde	secondary	1.53e-06	8.92e-04
Aluminum Prod.	Alcohols	secondary	1.50e-06	8.76e-04
Ferrite mfg.	Xylene (mixed isomers)	secondary	1.43e-06	8.39e-04
Steel Prod., cold-rolled, semi-finished	Phenol	secondary	1.43e-06	8.35e-04
ABS Production	Ethanol	secondary	1.40e-06	8.17e-04
Aluminum Prod.	Benzene	secondary	1.22e-06	7.14e-04
Invar	Acetylene	secondary	1.11e-06	6.50e-04
Ferrite mfg.	Acetylene	secondary	1.08e-06	6.34e-04
Invar	Ethylbenzene	secondary	9.65e-07	5.64e-04
Invar	Benzo[a]pyrene	secondary	9.36e-07	5.48e-04
Steel Prod., cold-rolled, semi-finished	Heptane	secondary	8.81e-07	5.15e-04
Lead	Benzene	secondary	8.32e-07	4.87e-04
Invar	Aromatic hydrocarbons	secondary	7.76e-07	4.54e-04
Invar	Methanol	secondary	6.92e-07	4.05e-04
Invar	Ethanol	secondary	6.13e-07	3.58e-04
Invar	Formaldehyde	secondary	5.89e-07	3.44e-04
Ferrite mfg.	Formaldehyde	secondary	5.74e-07	3.36e-04
Invar	Hexane	secondary	5.36e-07	3.13e-04
Ferrite mfg.	Hexane	secondary	5.23e-07	3.06e-04
Invar	Acetaldehyde	secondary	5.10e-07	2.98e-04
Steel Prod., cold-rolled, semi-finished	Aromatic hydrocarbons	secondary	5.05e-07	2.95e-04
Steel Prod., cold-rolled, semi-finished	Methanol	secondary	4.42e-07	2.59e-04
Lead	Ethylbenzene	secondary	4.11e-07	2.40e-04
Invar	Phenol	secondary	4.05e-07	2.37e-04
Ferrite mfg.	Phenol	secondary	3.95e-07	2.31e-04
Steel Prod., cold-rolled, semi-finished	Ethanol	secondary	3.56e-07	2.08e-04
Polystyrene Prod., high-impact	Polycyclic aromatic hydrocarbons	secondary	3.45e-07	2.02e-04
Steel Prod., cold-rolled, semi-finished	Acetaldehyde	secondary	2.96e-07	1.73e-04

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Invar	Heptane	secondary	2.67e-07	1.56e-04
Ferrite mfg.	Heptane	secondary	2.61e-07	1.53e-04
Steel Prod., cold-rolled, semi-finished	Ethylbenzene	secondary	2.35e-07	1.37e-04
Invar	Acetone	secondary	2.04e-07	1.19e-04
Polycarbonate Production	Ethanethiol	secondary	1.84e-07	1.07e-04
Ferrite mfg.	Aromatic hydrocarbons	secondary	1.62e-07	9.45e-05
Ferrite mfg.	Methanol	secondary	1.24e-07	7.28e-05
Invar	Aldehydes	secondary	1.21e-07	7.05e-05
Steel Prod., cold-rolled, semi-finished	Acetone	secondary	1.18e-07	6.91e-05
Ferrite mfg.	Aldehydes	secondary	1.14e-07	6.64e-05
Ferrite mfg.	Ethylbenzene	secondary	1.03e-07	6.03e-05
Ferrite mfg.	Ethanol	secondary	1.00e-07	5.86e-05
ABS Production	Aldehydes	secondary	9.38e-08	5.48e-05
Styrene-butadiene Copolymer Prod.	Aldehydes	secondary	9.16e-08	5.36e-05
Ferrite mfg.	Acetaldehyde	secondary	8.48e-08	4.96e-05
ABS Production	Ethanethiol	secondary	8.43e-08	4.93e-05
Lead	Ethanol	secondary	7.77e-08	4.54e-05
Steel Prod., cold-rolled, semi-finished	Benzo[a]pyrene	secondary	6.79e-08	3.97e-05
Lead	Acetaldehyde	secondary	6.57e-08	3.84e-05
Lead	Methanol	secondary	6.02e-08	3.52e-05
Lead	Aromatic hydrocarbons	secondary	5.16e-08	3.02e-05
Steel Prod., cold-rolled, semi-finished	Polycyclic aromatic hydrocarbons	secondary	3.72e-08	2.18e-05
Invar	Polycyclic aromatic hydrocarbons	secondary	3.63e-08	2.12e-05
Ferrite mfg.	Polycyclic aromatic hydrocarbons	secondary	3.54e-08	2.07e-05
Ferrite mfg.	Acetone	secondary	3.34e-08	1.95e-05
Lead	Acetone	secondary	2.57e-08	1.50e-05
Invar	Alcohols	secondary	2.38e-08	1.39e-05
Ferrite mfg.	Alcohols	secondary	2.33e-08	1.36e-05
Ferrite mfg.	Benzo[a]pyrene	secondary	1.88e-08	1.10e-05
Steel Prod., cold-rolled, semi-finished	Naphthalene	secondary	1.13e-08	6.62e-06
Aluminum Prod.	1,2-Dichlorotetrafluoroethane	secondary	5.02e-09	2.93e-06
Aluminum Prod.	Phenol	secondary	3.04e-09	1.78e-06
Lead	Aldehydes	secondary	2.15e-09	1.26e-06
Invar	1,2-Dichlorotetrafluoroethane	secondary	2.08e-09	1.22e-06
Lead	Benzo[a]pyrene	secondary	1.49e-09	8.73e-07
Invar	Trichlorofluoromethane	secondary	1.27e-10	7.40e-08
ABS Production	HCFC-22	secondary	5.59e-11	3.27e-08
Ferrite mfg.	Trichlorofluoromethane	secondary	4.71e-11	2.75e-08
Invar	Ethanethiol	secondary	3.47e-11	2.03e-08
Ferrite mfg.	Ethanethiol	secondary	3.38e-11	1.98e-08
Invar	Dichlorodifluoromethane	secondary	1.69e-11	9.85e-09
Lead	Acrolein	secondary	1.05e-11	6.17e-09
Lead	Benzaldehyde	secondary	5.52e-12	3.22e-09
Invar	Propionaldehyde	secondary	9.34e-14	5.46e-11
Steel Prod., cold-rolled, semi-finished	Propionaldehyde	secondary	9.34e-14	5.46e-11
Ferrite mfg.	Propionaldehyde	secondary	9.12e-14	5.33e-11
Steel Prod., cold-rolled, semi-finished	Benzaldehyde	secondary	2.51e-14	1.47e-11
Invar	Benzaldehyde	secondary	2.50e-14	1.46e-11
Ferrite mfg.	Benzaldehyde	secondary	2.43e-14	1.42e-11

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Invar	Acrolein	secondary	4.38e-15	2.56e-12
Ferrite mfg.	1,2-Dichlorotetrafluoroethane	secondary	1.54e-16	8.98e-14
Invar	HCFC-22	secondary	1.41e-18	8.22e-16
Ferrite mfg.	HCFC-22	secondary	1.37e-18	8.02e-16
Ferrite mfg.	Dichlorodifluoromethane	secondary	1.25e-18	7.30e-16
Total Materials Processing			4.69e-02	2.74e+01
Manufacturing Life-cycle Stage				
LPG Production	Hydrocarbons, remaining unspeciated	secondary	6.22e-02	3.64e+01
LPG Production	Nonmethane hydrocarbons, remaining unspeciated	secondary	4.24e-02	2.48e+01
LPG Production	Methane	secondary	5.87e-03	3.43e+00
LPG Production	Benzene	secondary	2.71e-03	1.59e+00
CRT tube mfg.	Toluene	primary	2.16e-03	1.26e+00
Natural Gas Prod.	Nonmethane hydrocarbons, remaining unspeciated	secondary	9.01e-04	5.27e-01
LPG Production	Aldehydes	secondary	6.61e-04	3.87e-01
Fuel Oil #6 Prod.	Hydrocarbons, remaining unspeciated	secondary	5.36e-04	3.13e-01
LPG Production	Formaldehyde	secondary	4.93e-04	2.88e-01
Fuel Oil #6 Prod.	Nonmethane hydrocarbons, remaining unspeciated	secondary	3.55e-04	2.08e-01
Natural Gas Prod.	Methane	secondary	3.51e-04	2.05e-01
LPG Production	Ethane	secondary	3.09e-04	1.81e-01
CRT tube mfg.	Xylene (mixed isomers)	primary	2.92e-04	1.71e-01
LPG Production	Pentane	secondary	2.65e-04	1.55e-01
Natural Gas Prod.	Benzene	secondary	2.39e-04	1.40e-01
LPG Production	Butane	secondary	2.09e-04	1.22e-01
Fuel Oil #2 Prod.	Hydrocarbons, remaining unspeciated	secondary	1.97e-04	1.15e-01
LPG Production	Hexane	secondary	1.88e-04	1.10e-01
Fuel Oil #2 Prod.	Nonmethane hydrocarbons, remaining unspeciated	secondary	1.34e-04	7.84e-02
US electric grid	Methane	model/secondary	7.20e-05	4.21e-02
CRT tube mfg.	Nonmethane hydrocarbons, remaining unspeciated	primary	6.33e-05	3.70e-02
Fuel Oil #6 Prod.	Methane	secondary	4.12e-05	2.41e-02
Natural Gas Prod.	Hydrocarbons, remaining unspeciated	secondary	2.48e-05	1.45e-02
Fuel Oil #4 Prod.	Hydrocarbons, remaining unspeciated	secondary	2.16e-05	1.26e-02
Fuel Oil #2 Prod.	Methane	secondary	1.80e-05	1.05e-02
PWB Mfg.	Formaldehyde	model/secondary	1.63e-05	9.55e-03
Fuel Oil #6 Prod.	Benzene	secondary	1.55e-05	9.06e-03
Fuel Oil #4 Prod.	Nonmethane hydrocarbons, remaining unspeciated	secondary	1.45e-05	8.47e-03
Fuel Oil #2 Prod	Benzene	secondary	8 06e-06	4 71e-03
Fuel Oil #6 Prod	Formaldehyde	secondary	5.03e-06	2.94e-03
Natural Gas Prod	Aldehydes	secondary	4 77e-06	2.79e-03
Japanese Electric Grid	Formaldehyde	model/secondary	4 30e-06	2.51e-03
Fuel Oil #6 Prod	Aldehydes	secondary	3 77e-06	2.20e-03
LPG Production	Isonhorone	secondary	2 95e-06	1.73e-03
Natural Gas Prod	Fthane	secondary	2.55 c 00	1.63e-03
Natural Gas Prod	Pentane	secondary	2.76e 00 2.39e-06	1.05e 03
LPG Production	Toluene	secondary	2.09e-06	1.700 03
Fuel Oil #2 Prod	Aldehydes	secondary	1.96e-06	1.15e-03
Natural Gas Prod	Butane	secondary	1.88e-06	1.10e-03
Fuel Oil #4 Prod	Methane	secondary	1.836-00	1.07e-03
Fuel Oil #6 Prod	Ethane	secondary	1.76-06	1.03e-03
LPG Production	Acetaldehyde	secondary	1.700-00	9.88e_04
		Secondary	1.070-00	2.000-04

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Natural Gas Prod.	Hexane	secondary	1.69e-06	9.87e-04
Fuel Oil #2 Prod.	Formaldehyde	secondary	1.62e-06	9.47e-04
LPG Production	Propionaldehyde	secondary	1.53e-06	8.96e-04
Fuel Oil #6 Prod.	Pentane	secondary	1.52e-06	8.87e-04
LPG Production	Methyl ethyl ketone	secondary	1.23e-06	7.22e-04
Fuel Oil #6 Prod.	Butane	secondary	1.20e-06	6.99e-04
Fuel Oil #6 Prod.	Hexane	secondary	1.07e-06	6.27e-04
Fuel Oil #2 Prod.	Ethane	secondary	9.17e-07	5.36e-04
LPG Production	Acrolein	secondary	8.60e-07	5.03e-04
Japanese Electric Grid	Toluene	model/secondary	8.05e-07	4.71e-04
Fuel Oil #2 Prod.	Pentane	secondary	7.89e-07	4.61e-04
LPG Production	o-xylene	secondary	7.85e-07	4.59e-04
Fuel Oil #4 Prod.	Benzene	secondary	7.62e-07	4.46e-04
LPG Production	n-Propane	secondary	7.01e-07	4.10e-04
US electric grid	Isophorone	model/secondary	6.32e-07	3.69e-04
Fuel Oil #2 Prod.	Butane	secondary	6.21e-07	3.63e-04
Japanese Electric Grid	Methane	model/secondary	5.69e-07	3.33e-04
US electric grid	Formaldehyde	model/secondary	5.69e-07	3.33e-04
Fuel Oil #2 Prod.	Hexane	secondary	5.57e-07	3.26e-04
Japanese Electric Grid	Isophorone	model/secondary	5.05e-07	2.95e-04
LPG Production	Ethylbenzene	secondary	4.83e-07	2.83e-04
LPG Production	Bromomethane	secondary	4.26e-07	2.49e-04
US electric grid	Acetaldehyde	model/secondary	3.61e-07	2.11e-04
US electric grid	Benzene	model/secondary	3.52e-07	2.06e-04
US electric grid	Propionaldehyde	model/secondary	3.28e-07	1.92e-04
Japanese Electric Grid	Acetaldehyde	model/secondary	2.89e-07	1.69e-04
Japanese Electric Grid	Benzene	model/secondary	2.88e-07	1.68e-04
LPG Production	Naphthalene	secondary	2.66e-07	1.55e-04
US electric grid	Methyl ethyl ketone	model/secondary	2.64e-07	1.54e-04
Japanese Electric Grid	Propionaldehyde	model/secondary	2.62e-07	1.53e-04
US electric grid	Toluene	model/secondary	2.28e-07	1.34e-04
Japanese Electric Grid	Methyl ethyl ketone	model/secondary	2.11e-07	1.23e-04
Fuel Oil #4 Prod.	Formaldehyde	secondary	1.89e-07	1.10e-04
Fuel Oil #4 Prod.	Aldehydes	secondary	1.86e-07	1.09e-04
US electric grid	Acrolein	model/secondary	1.84e-07	1.08e-04
Japanese Electric Grid	Naphthalene	model/secondary	1.68e-07	9.84e-05
Japanese Electric Grid	Acrolein	model/secondary	1.47e-07	8.59e-05
LPG Production	Styrene	secondary	1.27e-07	7.44e-05
US electric grid	Ethylbenzene	model/secondary	1.03e-07	6.01e-05
Natural Gas Prod.	Formaldehyde	secondary	1.02e-07	5.95e-05
LPG Production	Benzyl chloride	secondary	9.84e-08	5.75e-05
LPG Production	Methyl tert-butyl ether	secondary	9.32e-08	5.45e-05
US electric grid	Bromomethane	model/secondary	9.12e-08	5.33e-05
Japanese Electric Grid	Ethylbenzene	model/secondary	9.03e-08	5.28e-05
Fuel Oil #4 Prod.	Ethane	secondary	8.68e-08	5.07e-05
LPG Production	Phenol	secondary	8.15e-08	4.76e-05
LPG Production	Acetophenone	secondary	7.64e-08	4.47e-05
Fuel Oil #4 Prod.	Pentane	secondary	7.46e-08	4.36e-05
LPG Production	Methyl chloride	secondary	7.45e-08	4.35e-05

 Table M-17. CRT LCIA Results for the Photochemical Smog Impact Category

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Japanese Electric Grid	Bromomethane	model/secondary	7.28e-08	4.26e-05
Fuel Oil #4 Prod.	Butane	secondary	5.88e-08	3.44e-05
Fuel Oil #4 Prod.	Hexane	secondary	5.27e-08	3.08e-05
Natural Gas Prod.	o-xylene	secondary	5.27e-08	3.08e-05
LPG Production	Dichloromethane	secondary	4.07e-08	2.38e-05
US electric grid	Xylene (mixed isomers)	model/secondary	4.03e-08	2.36e-05
LPG Production	Aromatic hydrocarbons	secondary	4.00e-08	2.34e-05
US electric grid	Hexane	model/secondary	3.99e-08	2.33e-05
Japanese Electric Grid	Xylene (mixed isomers)	model/secondary	3.22e-08	1.88e-05
Japanese Electric Grid	Hexane	model/secondary	3.19e-08	1.86e-05
Natural Gas Prod.	Toluene	secondary	2.91e-08	1.70e-05
US electric grid	Styrene	model/secondary	2.72e-08	1.59e-05
LPG Production	Cumene	secondary	2.70e-08	1.58e-05
Fuel Oil #6 Prod.	Isophorone	secondary	2.38e-08	1.39e-05
US electric grid	Naphthalene	model/secondary	2.22e-08	1.30e-05
Japanese Electric Grid	Styrene	model/secondary	2.17e-08	1.27e-05
US electric grid	Benzyl chloride	model/secondary	2.10e-08	1.23e-05
US electric grid	Methyl tert-butyl ether	model/secondary	1.99e-08	1.17e-05
US electric grid	Phenol	model/secondary	1.74e-08	1.02e-05
LPG Production	Phenanthrene	secondary	1.72e-08	1.01e-05
Japanese Electric Grid	Benzyl chloride	model/secondary	1.68e-08	9.82e-06
US electric grid	Acetophenone	model/secondary	1.63e-08	9.55e-06
US electric grid	Methyl chloride	model/secondary	1.59e-08	9.32e-06
Japanese Electric Grid	Methyl tert-butyl ether	model/secondary	1.59e-08	9.28e-06
Japanese Electric Grid	o-xylene	model/secondary	1.47e-08	8.59e-06
Fuel Oil #6 Prod.	Toluene	secondary	1.41e-08	8.25e-06
Japanese Electric Grid	Phenol	model/secondary	1.39e-08	8.14e-06
Fuel Oil #6 Prod.	Acetaldehyde	secondary	1.36e-08	7.97e-06
Japanese Electric Grid	Acetophenone	model/secondary	1.30e-08	7.63e-06
Japanese Electric Grid	Methyl chloride	model/secondary	1.27e-08	7.44e-06
Fuel Oil #6 Prod.	Propionaldehyde	secondary	1.24e-08	7.23e-06
LPG Production	Vinyl acetate	secondary	1.13e-08	6.63e-06
Fuel Oil #6 Prod.	Methyl ethyl ketone	secondary	9.96e-09	5.82e-06
Fuel Oil #2 Prod.	Isophorone	secondary	9.26e-09	5.42e-06
US electric grid	Dichloromethane	model/secondary	8.72e-09	5.10e-06
LPG Production	Biphenyl	secondary	8.66e-09	5.06e-06
LPG Production	Chloroform	secondary	8.29e-09	4.85e-06
Japanese Electric Grid	Dichloromethane	model/secondary	6.96e-09	4.07e-06
Fuel Oil #6 Prod.	Acrolein	secondary	6.94e-09	4.06e-06
Fuel Oil #2 Prod.	Toluene	secondary	6.37e-09	3.72e-06
LPG Production	1,4-Dichlorobenzene	secondary	6.31e-09	3.69e-06
LPG Production	Tetrachloroethylene	secondary	6.04e-09	3.53e-06
LPG Production	Ethyl Chloride	secondary	5.90e-09	3.45e-06
US electric grid	Cumene	model/secondary	5.77e-09	3.38e-06
LPG Production	1,2-Dichloroethane	secondary	5.62e-09	3.29e-06
Fuel Oil #6 Prod.	n-Propane	secondary	5.62e-09	3.29e-06
Fuel Oil #2 Prod.	Acetaldehyde	secondary	5.30e-09	3.10e-06
LPG Production	Fluorene	secondary	5.28e-09	3.09e-06
Natural Gas Prod.	Isophorone	secondary	5.12e-09	3.00e-06

Fuel Oil #6 Prod.o-xylenesecondary4.91e-092.87e-06Fuel Oil #2 Prod.Propionaldehydesecondary4.81e-092.81e-06Japanese Electric GridCumene hydroperoxidemodel/secondary4.61e-092.69e-06LPG Production2-Methylnaphthalenesecondary4.57e-092.67e-06LPG ProductionFluoranthenesecondary4.31e-092.52e-06LPG ProductionAcenaphthenesecondary3.99e-092.33e-06Japanese Electric GridPhenanthrenemodel/secondary3.99e-092.32e-06Fuel Oil #6 Prod.Ethylbenzenesecondary3.89e-092.27e-06Fuel Oil #2 Prod.Methyl ethyl ketonesecondary3.87e-092.26e-06Fuel Oil #6 Prod.Bromomethanesecondary3.87e-092.26e-06Fuel Oil #6 Prod.Bromomethanesecondary3.29e-091.92e-06Japanese Electric GridAcenaphthenemodel/secondary3.29e-091.92e-06LPG ProductionEthylene dibromidesecondary3.20e-091.87e-06LPG ProductionChlorobenzenesecondary3.00e-091.81e-06US electric gridPhenanthrenemodel/secondary3.04e-091.78e-06Natural Gas Prod.Acetaldehydesecondary2.93e-091.71e-06LPG Production1,1,1-Trichloroethanesecondary2.81e-091.64e-06LPG ProductionPyrenesecondary2.81e-091.64e-06
Fuel Oil #2 Prod.Propionaldehydesecondary4.81e-092.81e-06Japanese Electric GridCumene hydroperoxidemodel/secondary4.61e-092.69e-06LPG Production2-Methylnaphthalenesecondary4.57e-092.67e-06LPG ProductionFluoranthenesecondary4.31e-092.52e-06LPG ProductionAcenaphthenesecondary3.99e-092.33e-06Japanese Electric GridPhenanthrenemodel/secondary3.96e-092.32e-06Fuel Oil #6 Prod.Ethylbenzenesecondary3.89e-092.27e-06Fuel Oil #2 Prod.Methyl ethyl ketonesecondary3.87e-092.26e-06Fuel Oil #6 Prod.Bromomethanesecondary3.87e-092.26e-06Fuel Oil #6 Prod.Bromomethanesecondary3.29e-091.92e-06Japanese Electric GridAcenaphthenemodel/secondary3.29e-091.92e-06Japanese Electric GridAcenaphthenemodel/secondary3.29e-091.92e-06LPG ProductionEthylene dibromidesecondary3.09e-091.87e-06LPG ProductionChlorobenzenesecondary3.09e-091.81e-06US electric gridPhenanthrenemodel/secondary3.09e-091.78e-06Natural Gas Prod.Acetaldehydesecondary2.93e-091.71e-06LPG Production1,1,1-Trichloroethanesecondary2.81e-091.64e-06LPG ProductionPyrenesecondary2.74e-091.60e-06
Japanese Electric GridCumene hydroperoxidemodel/secondary4.61e-092.69e-06LPG Production2-Methylnaphthalenesecondary4.57e-092.67e-06LPG ProductionFluoranthenesecondary4.31e-092.52e-06LPG ProductionAcenaphthenesecondary3.99e-092.33e-06Japanese Electric GridPhenanthrenemodel/secondary3.96e-092.32e-06Fuel Oil #6 Prod.Ethylbenzenesecondary3.89e-092.26e-06Fuel Oil #2 Prod.Methyl ethyl ketonesecondary3.87e-092.26e-06Fuel Oil #6 Prod.Bromomethanesecondary3.44e-092.01e-06Japanese Electric GridAcenaphthenemodel/secondary3.29e-091.92e-06LPG ProductionEthylene dibromidesecondary3.20e-091.87e-06LPG ProductionChlorobenzenesecondary3.09e-091.81e-06US electric gridPhenanthrenemodel/secondary3.04e-091.78e-06LPG Production1,1,1-Trichloroethanesecondary2.93e-091.71e-06LPG Production1,1,1-Trichloroethanesecondary2.81e-091.64e-06LPG ProductionPyrenesecondary2.81e-091.64e-06
LPG Production2-Methylnaphthalenesecondary4.57e-092.67e-06LPG ProductionFluoranthenesecondary4.31e-092.52e-06LPG ProductionAcenaphthenesecondary3.99e-092.33e-06Japanese Electric GridPhenanthrenemodel/secondary3.96e-092.32e-06Fuel Oil #6 Prod.Ethylbenzenesecondary3.89e-092.27e-06Fuel Oil #2 Prod.Methyl ethyl ketonesecondary3.87e-092.26e-06Fuel Oil #6 Prod.Bromomethanesecondary3.44e-092.01e-06Japanese Electric GridAcenaphthenemodel/secondary3.29e-091.92e-06LPG ProductionEthylene dibromidesecondary3.20e-091.87e-06LPG ProductionChlorobenzenesecondary3.09e-091.87e-06US electric gridPhenanthrenemodel/secondary3.09e-091.81e-06US electric gridPhenanthrenemodel/secondary3.04e-091.78e-06Natural Gas Prod.Acetaldehydesecondary2.93e-091.71e-06LPG Production1,1,1-Trichloroethanesecondary2.81e-091.64e-06LPG ProductionPyrenesecondary2.81e-091.64e-06LPG ProductionPyrenesecondary2.74e-091.60e-06
LPG ProductionFluoranthenesecondary4.31e-092.52e-06LPG ProductionAcenaphthenesecondary3.99e-092.33e-06Japanese Electric GridPhenanthrenemodel/secondary3.96e-092.32e-06Fuel Oil #6 Prod.Ethylbenzenesecondary3.89e-092.27e-06Fuel Oil #2 Prod.Methyl ethyl ketonesecondary3.87e-092.26e-06Fuel Oil #6 Prod.Bromomethanesecondary3.44e-092.01e-06Japanese Electric GridAcenaphthenemodel/secondary3.29e-091.92e-06LPG ProductionEthylene dibromidesecondary3.20e-091.87e-06LPG ProductionChlorobenzenesecondary3.09e-091.87e-06US electric gridPhenanthrenemodel/secondary3.04e-091.78e-06Natural Gas Prod.Acetaldehydesecondary2.93e-091.71e-06LPG Production1,1,1-Trichloroethanesecondary2.81e-091.64e-06LPG Production9yrenesecondary2.81e-091.64e-06
LPG ProductionAcenaphthenesecondary3.99e-092.33e-06Japanese Electric GridPhenanthrenemodel/secondary3.96e-092.32e-06Fuel Oil #6 Prod.Ethylbenzenesecondary3.89e-092.27e-06Fuel Oil #2 Prod.Methyl ethyl ketonesecondary3.87e-092.26e-06Fuel Oil #6 Prod.Bromomethanesecondary3.44e-092.01e-06Japanese Electric GridAcenaphthenemodel/secondary3.29e-091.92e-06LPG ProductionEthylene dibromidesecondary3.20e-091.87e-06LPG ProductionChlorobenzenesecondary3.09e-091.81e-06US electric gridPhenanthrenemodel/secondary3.04e-091.78e-06Natural Gas Prod.Acetaldehydesecondary2.93e-091.71e-06LPG Production1,1,1-Trichloroethanesecondary2.81e-091.64e-06LPG ProductionPyrenesecondary2.81e-091.64e-06
Japanese Electric GridPhenanthrenemodel/secondary3.96e-092.32e-06Fuel Oil #6 Prod.Ethylbenzenesecondary3.89e-092.27e-06Fuel Oil #2 Prod.Methyl ethyl ketonesecondary3.87e-092.26e-06Fuel Oil #6 Prod.Bromomethanesecondary3.44e-092.01e-06Japanese Electric GridAcenaphthenemodel/secondary3.29e-091.92e-06LPG ProductionEthylene dibromidesecondary3.20e-091.87e-06LPG ProductionChlorobenzenesecondary3.09e-091.81e-06US electric gridPhenanthrenemodel/secondary3.04e-091.78e-06Natural Gas Prod.Acetaldehydesecondary2.93e-091.71e-06LPG Production1,1,1-Trichloroethanesecondary2.81e-091.64e-06LPG ProductionPyrenesecondary2.81e-091.64e-06
Fuel Oil #6 Prod.Ethylbenzenesecondary3.89e-092.27e-06Fuel Oil #2 Prod.Methyl ethyl ketonesecondary3.87e-092.26e-06Fuel Oil #6 Prod.Bromomethanesecondary3.44e-092.01e-06Japanese Electric GridAcenaphthenemodel/secondary3.29e-091.92e-06LPG ProductionEthylene dibromidesecondary3.20e-091.87e-06LPG ProductionChlorobenzenesecondary3.09e-091.81e-06US electric gridPhenanthrenemodel/secondary3.04e-091.78e-06Natural Gas Prod.Acetaldehydesecondary2.93e-091.71e-06LPG Production1,1,1-Trichloroethanesecondary2.81e-091.64e-06LPG ProductionPyrenesecondary2.81e-091.64e-06
Fuel Oil #2 Prod.Methyl ethyl ketonesecondary3.87e-092.26e-06Fuel Oil #6 Prod.Bromomethanesecondary3.44e-092.01e-06Japanese Electric GridAcenaphthenemodel/secondary3.29e-091.92e-06LPG ProductionEthylene dibromidesecondary3.20e-091.87e-06LPG ProductionChlorobenzenesecondary3.09e-091.81e-06US electric gridPhenanthrenemodel/secondary3.04e-091.78e-06Natural Gas Prod.Acetaldehydesecondary2.93e-091.71e-06LPG Production1,1,1-Trichloroethanesecondary2.81e-091.64e-06LPG ProductionPyrenesecondary1.64e-061.64e-06
Fuel Oil #6 Prod.Bromomethanesecondary3.44e-092.01e-06Japanese Electric GridAcenaphthenemodel/secondary3.29e-091.92e-06LPG ProductionEthylene dibromidesecondary3.20e-091.87e-06LPG ProductionChlorobenzenesecondary3.09e-091.81e-06US electric gridPhenanthrenemodel/secondary3.04e-091.78e-06Natural Gas Prod.Acetaldehydesecondary2.93e-091.71e-06LPG Production1,1,1-Trichloroethanesecondary2.81e-091.64e-06LPG ProductionPyrenesecondary2.74e-091.60e-06
Japanese Electric GridAcenaphthenemodel/secondary3.29e-091.92e-06LPG ProductionEthylene dibromidesecondary3.20e-091.87e-06LPG ProductionChlorobenzenesecondary3.09e-091.81e-06US electric gridPhenanthrenemodel/secondary3.04e-091.78e-06Natural Gas Prod.Acetaldehydesecondary2.93e-091.71e-06LPG Production1,1,1-Trichloroethanesecondary2.81e-091.64e-06LPG ProductionPyrenesecondary2.74e-091.60e-06
LPG ProductionEthylene dibromidesecondary3.20e-091.87e-06LPG ProductionChlorobenzenesecondary3.09e-091.81e-06US electric gridPhenanthrenemodel/secondary3.04e-091.78e-06Natural Gas Prod.Acetaldehydesecondary2.93e-091.71e-06LPG Production1,1,1-Trichloroethanesecondary2.81e-091.64e-06LPG ProductionPyrenesecondary2.74e-091.60e-06
LPG ProductionChlorobenzenesecondary3.09e-091.81e-06US electric gridPhenanthrenemodel/secondary3.04e-091.78e-06Natural Gas Prod.Acetaldehydesecondary2.93e-091.71e-06LPG Production1,1,1-Trichloroethanesecondary2.81e-091.64e-06LPG ProductionPyrenesecondary2.74e-091.60e-06
US electric gridPhenanthrenemodel/secondary3.04e-091.78e-06Natural Gas Prod.Acetaldehydesecondary2.93e-091.71e-06LPG Production1,1,1-Trichloroethanesecondary2.81e-091.64e-06LPG ProductionPyrenesecondary2.74e-091.60e-06
Natural Gas Prod.Acetaldehydesecondary2.93e-091.71e-06LPG Production1,1,1-Trichloroethanesecondary2.81e-091.64e-06LPG ProductionPyrenesecondary2.74e-091.60e-06
LPG Production1,1,1-Trichloroethanesecondary2.81e-091.64e-06LPG ProductionPyrenesecondary2.74e-091.60e-06
LPG Production Pyrene secondary 2.74e-09 1.60e-06
Fuel Oil #2 Prod.Acroleinsecondary2.70e-091.58e-06
Natural Gas Prod.Propionaldehydesecondary2.66e-091.55e-06
US electric grid Vinyl acetate model/secondary 2.43e-09 1.42e-06
Fuel Oil #2 Prod.o-xylenesecondary2.36e-091.38e-06
Fuel Oil #2 Prod.n-Propanesecondary2.20e-091.28e-06
Natural Gas Prod. Methyl ethyl ketone secondary 2.14e-09 1.25e-06
Japanese Electric Grid Vinyl acetate model/secondary 1.93e-09 1.13e-06
US electric grid Biphenyl model/secondary 1.85e-09 1.08e-06
US electric grid Chloroform model/secondary 1.77e-09 1.04e-06
LPG Production Acenaphthylene secondary 1.63e-09 9.52e-07
Fuel Oil #6 Prod.Naphthalenesecondary1.61e-099.41e-07
LPG Production Anthracene secondary 1.59e-09 9.28e-07
Fuel Oil #2 Prod.Ethylbenzenesecondary1.52e-098.86e-07
Natural Gas Prod. Acrolein secondary 1.49e-09 8.72e-07
Japanese Electric Grid Biphenyl model/secondary 1.48e-09 8.64e-07
LPG Production 2,4-Dinitrotoluene secondary 1.43e-09 8.34e-07
Japanese Electric Grid Chloroform model/secondary 1.42e-09 8.28e-07
Japanese Electric Grid Fluorene model/secondary 1.39e-09 8.14e-07
Japanese Electric Grid 1,1,1-Trichloroethane model/secondary 1.36e-09 7.94e-07
Natural Gas Prod. Naphthalene secondary 1.34e-09 7.84e-07
Natural Gas Prod. n-Propane secondary 1.34e-09 7.83e-07
Fuel Oil #2 Prod.Bromomethanesecondary1.34e-097.82e-07
Japanese Electric Grid Fluoranthene model/secondary 1.33e-09 7.77e-07
US electric grid Tetrachloroethylene model/secondary 1.29e-09 7.56e-07
US electric grid Ethyl Chloride model/secondary 1.26e-09 7.38e-07
US electric grid 1.2-Dichloroethane model/secondary 1.20e-09 7.03e-07
Japanese Electric Grid Tetrachloroethylene model/secondary 1.03e-09 6.03e-07
Fuel Oil #6 Prod. Styrene secondary 1 03e-09 6 01e-07
US electric grid Fluorene model/secondary 1 02e-09 5 96e-07
Japanese Electric Grid Ethyl Chloride model/secondary 1 01e-09 5 89e-07
Fuel Oil #4 Prod. Isophorone secondary 9 88e-10 5 78e-07
LPG Production 2-Chloroacetophenone secondary 9.84e-10 5.75e-07

 Table M-17. CRT LCIA Results for the Photochemical Smog Impact Category

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Japanese Electric Grid	1,2-Dichloroethane	model/secondary	9.60e-10	5.61e-07
Japanese Electric Grid	Pyrene	model/secondary	9.59e-10	5.61e-07
LPG Production	Benzo[a]anthracene	secondary	9.50e-10	5.55e-07
LPG Production	Chrysene	secondary	9.11e-10	5.33e-07
Natural Gas Prod.	Ethylbenzene	secondary	8.40e-10	4.91e-07
US electric grid	Fluoranthene	model/secondary	8.15e-10	4.77e-07
Fuel Oil #2 Prod.	Naphthalene	secondary	7.96e-10	4.65e-07
Fuel Oil #6 Prod.	Benzyl chloride	secondary	7.94e-10	4.64e-07
Fuel Oil #6 Prod.	Methyl tert-butyl ether	secondary	7.52e-10	4.40e-07
Natural Gas Prod.	Bromomethane	secondary	7.39e-10	4.32e-07
LPG Production	Indeno(1,2,3-cd)pyrene	secondary	7.07e-10	4.13e-07
US electric grid	o-xylene	model/secondary	6.93e-10	4.05e-07
US electric grid	Acenaphthene	model/secondary	6.90e-10	4.03e-07
US electric grid	Ethylene dibromide	model/secondary	6.84e-10	4.00e-07
US electric grid	Chlorobenzene	model/secondary	6.61e-10	3.87e-07
Fuel Oil #6 Prod.	Phenol	secondary	6.57e-10	3.84e-07
US electric grid	1,1,1-Trichloroethane	model/secondary	6.43e-10	3.76e-07
Fuel Oil #4 Prod.	Toluene	secondary	6.37e-10	3.72e-07
Fuel Oil #6 Prod.	Acetophenone	secondary	6.16e-10	3.60e-07
Japanese Electric Grid	Benzo[a]anthracene	model/secondary	6.09e-10	3.56e-07
Fuel Oil #6 Prod.	Methyl chloride	secondary	6.01e-10	3.51e-07
Fuel Oil #4 Prod.	Acetaldehyde	secondary	5.65e-10	3.30e-07
LPG Production	Benzo[b,j,k]fluoranthene	secondary	5.60e-10	3.27e-07
Japanese Electric Grid	Ethylene dibromide	model/secondary	5.46e-10	3.19e-07
Japanese Electric Grid	Chlorobenzene	model/secondary	5.28e-10	3.09e-07
LPG Production	Benzo[a]pyrene	secondary	5.20e-10	3.04e-07
Fuel Oil #4 Prod.	Propionaldehyde	secondary	5.13e-10	3.00e-07
LPG Production	Benzo[g,h,i]perylene	secondary	4.22e-10	2.47e-07
Fuel Oil #4 Prod.	Methyl ethyl ketone	secondary	4.13e-10	2.41e-07
Japanese Electric Grid	Chrysene	model/secondary	4.07e-10	2.38e-07
US electric grid	Pyrene	model/secondary	4.05e-10	2.37e-07
Fuel Oil #2 Prod.	Styrene	secondary	3.99e-10	2.33e-07
Japanese Electric Grid	Anthracene	model/secondary	3.47e-10	2.03e-07
Japanese Electric Grid	Indeno(1,2,3-cd)pyrene	model/secondary	3.41e-10	1.99e-07
Fuel Oil #6 Prod.	Dichloromethane	secondary	3.29e-10	1.92e-07
Japanese Electric Grid	Benzo[g,h,i]perylene	model/secondary	3.27e-10	1.91e-07
Fuel Oil #2 Prod.	Benzyl chloride	secondary	3.09e-10	1.80e-07
US electric grid	2,4-Dinitrotoluene	model/secondary	3.05e-10	1.78e-07
Japanese Electric Grid	Benzo[b,j,k]fluoranthene	model/secondary	2.96e-10	1.73e-07
Fuel Oil #2 Prod.	Methyl tert-butyl ether	secondary	2.92e-10	1.71e-07
Fuel Oil #4 Prod.	Acrolein	secondary	2.88e-10	1.68e-07
US electric grid	Acenaphthylene	model/secondary	2.74e-10	1.60e-07
Fuel Oil #2 Prod.	Phenol	secondary	2.56e-10	1.49e-07
Japanese Electric Grid	Acenaphthylene	model/secondary	2.51e-10	1.47e-07
Japanese Electric Grid	2,4-Dinitrotoluene	model/secondary	2.43e-10	1.42e-07
Fuel Oil #2 Prod.	Acetophenone	secondary	2.40e-10	1.40e-07
US electric grid	Anthracene	model/secondary	2.37e-10	1.38e-07
Fuel Oil #4 Prod.	n-Propane	secondary	2.34e-10	1.37e-07
Fuel Oil #2 Prod.	Methyl chloride	secondary	2.34e-10	1.37e-07

 Table M-17. CRT LCIA Results for the Photochemical Smog Impact Category

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Fuel Oil #4 Prod.	o-xylene	secondary	2.30e-10	1.35e-07
Natural Gas Prod.	Styrene	secondary	2.21e-10	1.29e-07
Fuel Oil #6 Prod.	Cumene	secondary	2.18e-10	1.27e-07
US electric grid	2-Chloroacetophenone	model/secondary	2.10e-10	1.23e-07
Japanese Electric Grid	2-Methylnaphthalene	model/secondary	1.81e-10	1.06e-07
Natural Gas Prod.	Benzyl chloride	secondary	1.71e-10	9.97e-08
Japanese Electric Grid	2-Chloroacetophenone	model/secondary	1.68e-10	9.83e-08
Natural Gas Prod.	Methyl tert-butyl ether	secondary	1.62e-10	9.45e-08
Fuel Oil #4 Prod.	Ethylbenzene	secondary	1.61e-10	9.43e-08
Fuel Oil #4 Prod.	Bromomethane	secondary	1.43e-10	8.33e-08
Natural Gas Prod.	Phenol	secondary	1.41e-10	8.26e-08
Fuel Oil #6 Prod.	Aromatic hydrocarbons	secondary	1.35e-10	7.90e-08
Natural Gas Prod.	Acetophenone	secondary	1.32e-10	7.75e-08
Fuel Oil #6 Prod.	Phenanthrene	secondary	1.30e-10	7.62e-08
US electric grid	Benzo[b,j,k]fluoranthene	model/secondary	1.29e-10	7.56e-08
Natural Gas Prod.	Methyl chloride	secondary	1.29e-10	7.55e-08
Fuel Oil #2 Prod.	Dichloromethane	secondary	1.28e-10	7.47e-08
US electric grid	Chrysene	model/secondary	1.24e-10	7.25e-08
US electric grid	Benzo[a]anthracene	model/secondary	1.13e-10	6.59e-08
Fuel Oil #2 Prod.	Aromatic hydrocarbons	secondary	1.12e-10	6.56e-08
LPG Production	5-Methyl chrysene	secondary	1.12e-10	6.55e-08
Fuel Oil #6 Prod.	Vinyl acetate	secondary	9.15e-11	5.35e-08
Fuel Oil #2 Prod.	Cumene	secondary	8.47e-11	4.95e-08
US electric grid	Indeno(1,2,3-cd)pyrene	model/secondary	8.01e-11	4.68e-08
Fuel Oil #4 Prod.	Naphthalene	secondary	7.67e-11	4.49e-08
Natural Gas Prod.	Dichloromethane	secondary	7.07e-11	4.13e-08
Fuel Oil #6 Prod.	Biphenyl	secondary	6.98e-11	4.08e-08
Fuel Oil #6 Prod.	Chloroform	secondary	6.69e-11	3.91e-08
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	5.68e-11	3.32e-08
Natural Gas Prod.	Phenanthrene	secondary	5.36e-11	3.14e-08
Fuel Oil #2 Prod.	Phenanthrene	secondary	5.35e-11	3.13e-08
Fuel Oil #6 Prod.	Tetrachloroethylene	secondary	4.87e-11	2.85e-08
Fuel Oil #6 Prod.	Ethyl Chloride	secondary	4.76e-11	2.78e-08
Natural Gas Prod.	Cumene	secondary	4.68e-11	2.74e-08
Fuel Oil #6 Prod.	1,2-Dichloroethane	secondary	4.53e-11	2.65e-08
LPG Production	Polycyclic aromatic hydrocarbons	secondary	4.44e-11	2.60e-08
US electric grid	Benzo[g,h,i]perylene	model/secondary	4.38e-11	2.56e-08
Fuel Oil #4 Prod.	Styrene	secondary	4.26e-11	2.49e-08
US electric grid	Benzo[a]pyrene	model/secondary	4.14e-11	2.42e-08
Natural Gas Prod.	2-Methylnaphthalene	secondary	4.12e-11	2.41e-08
Fuel Oil #6 Prod.	Fluorene	secondary	4.08e-11	2.39e-08
Fuel Oil #6 Prod.	1,4-Dichlorobenzene	secondary	3.60e-11	2.11e-08
Fuel Oil #2 Prod.	Vinyl acetate	secondary	3.56e-11	2.08e-08
Japanese Electric Grid	Benzo[a]pyrene	model/secondary	3.31e-11	1.93e-08
Fuel Oil #4 Prod.	Benzyl chloride	secondary	3.29e-11	1.92e-08
Fuel Oil #6 Prod.	Fluoranthene	secondary	3.28e-11	1.92e-08
US electric grid	2-Methylnaphthalene	model/secondary	3.24e-11	1.89e-08
Fuel Oil #4 Prod.	Methyl tert-butyl ether	secondary	3.12e-11	1.82e-08
Fuel Oil #4 Prod.	Phenol	secondary	2.73e-11	1.59e-08

 Table M-17. CRT LCIA Results for the Photochemical Smog Impact Category
Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Fuel Oil #2 Prod.	Biphenyl	secondary	2.72e-11	1.59e-08
Fuel Oil #6 Prod.	Acenaphthene	secondary	2.65e-11	1.55e-08
Fuel Oil #6 Prod.	2-Methylnaphthalene	secondary	2.61e-11	1.53e-08
Fuel Oil #2 Prod.	Chloroform	secondary	2.60e-11	1.52e-08
Fuel Oil #6 Prod.	Ethylene dibromide	secondary	2.58e-11	1.51e-08
Fuel Oil #4 Prod.	Acetophenone	secondary	2.56e-11	1.49e-08
Fuel Oil #6 Prod.	Chlorobenzene	secondary	2.49e-11	1.46e-08
Fuel Oil #4 Prod.	Methyl chloride	secondary	2.49e-11	1.46e-08
US electric grid	5-Methyl chrysene	model/secondary	2.40e-11	1.40e-08
Fuel Oil #6 Prod.	1,1,1-Trichloroethane	secondary	2.27e-11	1.33e-08
Natural Gas Prod.	Vinyl acetate	secondary	1.97e-11	1.15e-08
Fuel Oil #6 Prod.	Pyrene	secondary	1.94e-11	1.13e-08
Japanese Electric Grid	5-Methyl chrysene	model/secondary	1.91e-11	1.12e-08
Fuel Oil #2 Prod.	Tetrachloroethylene	secondary	1.90e-11	1.11e-08
Fuel Oil #2 Prod.	1,4-Dichlorobenzene	secondary	1.87e-11	1.10e-08
Fuel Oil #2 Prod.	Ethyl Chloride	secondary	1.85e-11	1.08e-08
Fuel Oil #2 Prod.	1,2-Dichloroethane	secondary	1.76e-11	1.03e-08
Fuel Oil #2 Prod.	Fluorene	secondary	1.64e-11	9.61e-09
Natural Gas Prod.	Biphenyl	secondary	1.50e-11	8.78e-09
Natural Gas Prod.	Chloroform	secondary	1.44e-11	8.41e-09
Fuel Oil #4 Prod.	Dichloromethane	secondary	1.36e-11	7.97e-09
Fuel Oil #2 Prod.	2-Methylnaphthalene	secondary	1.36e-11	7.94e-09
Fuel Oil #2 Prod.	Fluoranthene	secondary	1.34e-11	7.82e-09
Natural Gas Prod.	Fluorene	secondary	1.31e-11	7.66e-09
Fuel Oil #6 Prod.	Acenaphthylene	secondary	1.23e-11	7.18e-09
Fuel Oil #2 Prod.	Acenaphthene	secondary	1.21e-11	7.09e-09
Natural Gas Prod.	Pyrene	secondary	1.17e-11	6.86e-09
Natural Gas Prod.	Fluoranthene	secondary	1.17e-11	6.84e-09
Fuel Oil #6 Prod.	2,4-Dinitrotoluene	secondary	1.15e-11	6.73e-09
Fuel Oil #6 Prod.	Anthracene	secondary	1.14e-11	6.69e-09
Natural Gas Prod.	Tetrachloroethylene	secondary	1.05e-11	6.13e-09
Natural Gas Prod.	Ethyl Chloride	secondary	1.02e-11	5.98e-09
Natural Gas Prod.	Acenaphthene	secondary	1.01e-11	5.89e-09
Fuel Oil #2 Prod.	Ethylene dibromide	secondary	1.00e-11	5.86e-09
Natural Gas Prod.	1,2-Dichloroethane	secondary	9.75e-12	5.70e-09
Fuel Oil #2 Prod.	Chlorobenzene	secondary	9.70e-12	5.67e-09
Fuel Oil #4 Prod.	Aromatic hydrocarbons	secondary	9.11e-12	5.33e-09
Fuel Oil #4 Prod.	Cumene	secondary	9.03e-12	5.28e-09
Fuel Oil #2 Prod.	1,1,1-Trichloroethane	secondary	8.82e-12	5.15e-09
Fuel Oil #2 Prod.	Pyrene	secondary	8.40e-12	4.91e-09
Fuel Oil #6 Prod.	2-Chloroacetophenone	secondary	7.94e-12	4.64e-09
Fuel Oil #6 Prod.	Chrysene	secondary	6.27e-12	3.67e-09
Natural Gas Prod.	Anthracene	secondary	6.11e-12	3.57e-09
Fuel Oil #6 Prod.	Benzo[a]anthracene	secondary	5.93e-12	3.47e-09
Fuel Oil #4 Prod.	Phenanthrene	secondary	5.57e-12	3.26e-09
Natural Gas Prod.	Ethylene dibromide	secondary	5.54e-12	3.24e-09
Natural Gas Prod.	Chlorobenzene	secondary	5.36e-12	3.13e-09
Natural Gas Prod.	Acenaphthylene	secondary	5.32e-12	3.11e-09
Fuel Oil #2 Prod.	Acenaphthylene	secondary	5.05e-12	2.95e-09

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Fuel Oil #2 Prod.	Anthracene	secondary	4.88e-12	2.86e-09
Natural Gas Prod.	1,1,1-Trichloroethane	secondary	4.87e-12	2.85e-09
Fuel Oil #6 Prod.	Indeno(1,2,3-cd)pyrene	secondary	4.65e-12	2.72e-09
Fuel Oil #6 Prod.	Benzo[b,j,k]fluoranthene	secondary	4.52e-12	2.64e-09
Fuel Oil #2 Prod.	2,4-Dinitrotoluene	secondary	4.47e-12	2.62e-09
Natural Gas Prod.	Benzo[a]anthracene	secondary	4.27e-12	2.49e-09
Natural Gas Prod.	Chrysene	secondary	4.11e-12	2.40e-09
Fuel Oil #6 Prod.	Benzo[a]pyrene	secondary	3.88e-12	2.27e-09
Fuel Oil #4 Prod.	Vinyl acetate	secondary	3.79e-12	2.22e-09
Natural Gas Prod.	Indeno(1,2,3-cd)pyrene	secondary	3.75e-12	2.19e-09
Fuel Oil #2 Prod.	2-Chloroacetophenone	secondary	3.09e-12	1.80e-09
Fuel Oil #4 Prod.	Biphenyl	secondary	2.90e-12	1.69e-09
Fuel Oil #2 Prod.	Benzo[a]anthracene	secondary	2.86e-12	1.67e-09
Fuel Oil #2 Prod.	Chrysene	secondary	2.78e-12	1.63e-09
Fuel Oil #4 Prod.	Chloroform	secondary	2.77e-12	1.62e-09
Fuel Oil #6 Prod.	Benzo[g,h,i]perylene	secondary	2.61e-12	1.53e-09
Natural Gas Prod.	2,4-Dinitrotoluene	secondary	2.47e-12	1.45e-09
Natural Gas Prod.	Benzo[g,h,i]perylene	secondary	2.43e-12	1.42e-09
Natural Gas Prod.	Benzo[a]pyrene	secondary	2.40e-12	1.40e-09
Fuel Oil #2 Prod.	Indeno(1,2,3-cd)pyrene	secondary	2.14e-12	1.25e-09
Fuel Oil #4 Prod.	Tetrachloroethylene	secondary	2.02e-12	1.18e-09
Fuel Oil #4 Prod.	Ethyl Chloride	secondary	1.97e-12	1.15e-09
Fuel Oil #4 Prod.	1,2-Dichloroethane	secondary	1.88e-12	1.10e-09
Fuel Oil #4 Prod.	1,4-Dichlorobenzene	secondary	1.77e-12	1.04e-09
Fuel Oil #2 Prod.	Benzo[b,j,k]fluoranthene	secondary	1.76e-12	1.03e-09
Fuel Oil #4 Prod.	Fluorene	secondary	1.73e-12	1.01e-09
Natural Gas Prod.	2-Chloroacetophenone	secondary	1.71e-12	9.97e-10
Fuel Oil #2 Prod.	Benzo[a]pyrene	secondary	1.61e-12	9.41e-10
Fuel Oil #4 Prod.	Fluoranthene	secondary	1.40e-12	8.17e-10
Fuel Oil #4 Prod.	2-Methylnaphthalene	secondary	1.28e-12	7.51e-10
Fuel Oil #2 Prod.	Benzo[g,h,i]perylene	secondary	1.27e-12	7.42e-10
Fuel Oil #4 Prod.	Acenaphthene	secondary	1.21e-12	7.05e-10
Fuel Oil #4 Prod.	Ethylene dibromide	secondary	1.07e-12	6.25e-10
Fuel Oil #4 Prod.	Chlorobenzene	secondary	1.03e-12	6.05e-10
Natural Gas Prod.	Benzo[b,j,k]fluoranthene	secondary	9.72e-13	5.68e-10
Fuel Oil #4 Prod.	1,1,1-Trichloroethane	secondary	9.40e-13	5.50e-10
Fuel Oil #6 Prod.	5-Methyl chrysene	secondary	9.04e-13	5.28e-10
Fuel Oil #4 Prod.	Pyrene	secondary	8.54e-13	4.99e-10
Natural Gas Prod.	Aromatic hydrocarbons	secondary	8.08e-13	4.72e-10
Fuel Oil #4 Prod.	Acenaphthylene	secondary	5.25e-13	3.07e-10
Fuel Oil #4 Prod.	Anthracene	secondary	5.00e-13	2.92e-10
Fuel Oil #4 Prod.	2,4-Dinitrotoluene	secondary	4.77e-13	2.79e-10
Fuel Oil #2 Prod.	5-Methyl chrysene	secondary	3.51e-13	2.05e-10
Fuel Oil #4 Prod.	2-Chloroacetophenone	secondary	3.29e-13	1.92e-10
Fuel Oil #4 Prod.	Chrysene	secondary	2.80e-13	1.64e-10
Fuel Oil #4 Prod.	Benzo[a]anthracene	secondary	2.78e-13	1.63e-10
Fuel Oil #4 Prod.	Indeno(1,2,3-cd)pyrene	secondary	2.12e-13	1.24e-10
Natural Gas Prod.	5-Methyl chrysene	secondary	1.94e-13	1.14e-10
Fuel Oil #4 Prod.	Benzo[b,j,k]fluoranthene	secondary	1.87e-13	1.10e-10

 Table M-17. CRT LCIA Results for the Photochemical Smog Impact Category

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Fuel Oil #4 Prod.	Benzo[a]pyrene	secondary	1.67e-13	9.75e-11
Fuel Oil #6 Prod.	Polycyclic aromatic hydrocarbons	secondary	1.50e-13	8.78e-11
Fuel Oil #2 Prod.	Polycyclic aromatic hydrocarbons	secondary	1.25e-13	7.29e-11
Fuel Oil #4 Prod.	Benzo[g,h,i]perylene	secondary	1.23e-13	7.20e-11
Fuel Oil #4 Prod.	5-Methyl chrysene	secondary	3.75e-14	2.19e-11
Fuel Oil #4 Prod.	Polycyclic aromatic hydrocarbons	secondary	1.01e-14	5.92e-12
Natural Gas Prod.	Polycyclic aromatic hydrocarbons	secondary	8.98e-16	5.25e-13
Total Manufacturing			1.21e-01	7.07e+01
Use, Maintenance and Repair I	life-cycle Stage			
US electric grid	Methane	model/secondary	4.51e-03	2.64e+00
US electric grid	Isophorone	model/secondary	3.96e-05	2.31e-02
US electric grid	Formaldehyde	model/secondary	3.56e-05	2.08e-02
US electric grid	Acetaldehyde	model/secondary	2.26e-05	1.32e-02
US electric grid	Benzene	model/secondary	2.21e-05	1.29e-02
US electric grid	Propionaldehyde	model/secondary	2.05e-05	1.20e-02
US electric grid	Methyl ethyl ketone	model/secondary	1.65e-05	9.67e-03
US electric grid	Toluene	model/secondary	1.43e-05	8.37e-03
US electric grid	Acrolein	model/secondary	1.15e-05	6.74e-03
US electric grid	Ethylbenzene	model/secondary	6.44e-06	3.77e-03
US electric grid	Bromomethane	model/secondary	5.71e-06	3.34e-03
US electric grid	Xylene (mixed isomers)	model/secondary	2.53e-06	1.48e-03
US electric grid	Hexane	model/secondary	2.50e-06	1.46e-03
US electric grid	Styrene	model/secondary	1.71e-06	9.98e-04
US electric grid	Naphthalene	model/secondary	1.39e-06	8.14e-04
US electric grid	Benzyl chloride	model/secondary	1.32e-06	7.71e-04
US electric grid	Methyl tert-butyl ether	model/secondary	1.25e-06	7.30e-04
US electric grid	Phenol	model/secondary	1.09e-06	6.38e-04
US electric grid	Acetophenone	model/secondary	1.02e-06	5.99e-04
US electric grid	Methyl chloride	model/secondary	9.98e-07	5.84e-04
US electric grid	Dichloromethane	model/secondary	5.46e-07	3.19e-04
US electric grid	Cumene	model/secondary	3.62e-07	2.11e-04
US electric grid	Phenanthrene	model/secondary	1.91e-07	1.11e-04
US electric grid	Vinyl acetate	model/secondary	1.52e-07	8.89e-05
US electric grid	Biphenyl	model/secondary	1.16e-07	6.78e-05
US electric grid	Chloroform	model/secondary	1.11e-07	6.50e-05
US electric grid	Tetrachloroethylene	model/secondary	8.10e-08	4.73e-05
US electric grid	Ethyl Chloride	model/secondary	7.91e-08	4.62e-05
US electric grid	1,2-Dichloroethane	model/secondary	7.53e-08	4.40e-05
US electric grid	Fluorene	model/secondary	6.39e-08	3.74e-05
US electric grid	Fluoranthene	model/secondary	5.11e-08	2.99e-05
US electric grid	o-xylene	model/secondary	4.34e-08	2.54e-05
US electric grid	Acenaphthene	model/secondary	4.32e-08	2.53e-05
US electric grid	Ethylene dibromide	model/secondary	4.28e-08	2.50e-05
US electric grid	Chlorobenzene	model/secondary	4.14e-08	2.42e-05
US electric grid	1,1,1-Trichloroethane	model/secondary	4.03e-08	2.35e-05
US electric grid	Pyrene	model/secondary	2.53e-08	1.48e-05
US electric grid	2,4-Dinitrotoluene	model/secondary	1.91e-08	1.12e-05
US electric grid	Acenaphthylene	model/secondary	1.72e-08	1.00e-05
US electric grid	Anthracene	model/secondary	1.48e-08	8.66e-06

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
US electric grid	2-Chloroacetophenone	model/secondary	1.32e-08	7.71e-06
US electric grid	Benzo[b,j,k]fluoranthene	model/secondary	8.10e-09	4.73e-06
US electric grid	Chrysene	model/secondary	7.77e-09	4.54e-06
US electric grid	Benzo[a]anthracene	model/secondary	7.06e-09	4.13e-06
US electric grid	Indeno(1,2,3-cd)pyrene	model/secondary	5.02e-09	2.93e-06
US electric grid	Benzo[g,h,i]perylene	model/secondary	2.74e-09	1.60e-06
US electric grid	Benzo[a]pyrene	model/secondary	2.59e-09	1.52e-06
US electric grid	2-Methylnaphthalene	model/secondary	2.03e-09	1.19e-06
US electric grid	5-Methyl chrysene	model/secondary	1.50e-09	8.78e-07
Total Use, Maintenance and Re	pair		4.72e-03	2.76e+00
End-of-life Life-cycle Stage				
CRT landfilling	Hydrocarbons, remaining unspeciated	primary	1.07e-04	6.28e-02
LPG Production	Hydrocarbons, remaining unspeciated	secondary	5.37e-07	3.14e-04
US electric grid	Methane	model/secondary	4.52e-07	2.64e-04
CRT landfilling	Methane	primary	3.81e-07	2.23e-04
LPG Production	Nonmethane hydrocarbons, remaining unspeciated	secondary	3.66e-07	2.14e-04
CRT landfilling	Benzene	primary	1.28e-07	7.48e-05
LPG Production	Methane	secondary	5.07e-08	2.96e-05
CRT landfilling	Toluene	primary	2.69e-08	1.57e-05
LPG Production	Benzene	secondary	2.34e-08	1.37e-05
CRT landfilling	Xylene (mixed isomers)	primary	1.88e-08	1.10e-05
LPG Production	Aldehydes	secondary	5.71e-09	3.34e-06
LPG Production	Formaldehyde	secondary	4.25e-09	2.49e-06
US electric grid	Isophorone	model/secondary	3.96e-09	2.32e-06
CRT landfilling	Ethylbenzene	primary	3.75e-09	2.19e-06
US electric grid	Formaldehyde	model/secondary	3.56e-09	2.08e-06
LPG Production	Ethane	secondary	2.66e-09	1.56e-06
LPG Production	Pentane	secondary	2.29e-09	1.34e-06
US electric grid	Acetaldehyde	model/secondary	2.27e-09	1.32e-06
US electric grid	Benzene	model/secondary	2.21e-09	1.29e-06
US electric grid	Propionaldehyde	model/secondary	2.06e-09	1.20e-06
LPG Production	Butane	secondary	1.81e-09	1.06e-06
US electric grid	Methyl ethyl ketone	model/secondary	1.66e-09	9.68e-07
LPG Production	Hexane	secondary	1.62e-09	9.47e-07
US electric grid	Toluene	model/secondary	1.43e-09	8.37e-07
US electric grid	Acrolein	model/secondary	1.15e-09	6.74e-07
US electric grid	Ethylbenzene	model/secondary	6.44e-10	3.77e-07
US electric grid	Bromomethane	model/secondary	5.71e-10	3.34e-07
US electric grid	Xylene (mixed isomers)	model/secondary	2.53e-10	1.48e-07
US electric grid	Hexane	model/secondary	2.50e-10	1.46e-07
US electric grid	Styrene	model/secondary	1.71e-10	9.98e-08
US electric grid	Naphthalene	model/secondary	1.39e-10	8.14e-08
US electric grid	Benzyl chloride	model/secondary	1.32e-10	7.71e-08
US electric grid	Methyl tert-butyl ether	model/secondary	1.25e-10	7.31e-08
US electric grid	Phenol	model/secondary	1.09e-10	6.39e-08
US electric grid	Acetophenone	model/secondary	1.02e-10	5.99e-08
US electric grid	Methyl chloride	model/secondary	9.99e-11	5.84e-08
CRT Incineration	Trichloroethylene	secondary	9.69e-11	5.67e-08
CRT landfilling	Trichloroethylene	primary	9.04e-11	5.28e-08

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
CRT Incineration	Vinyl chloride	secondary	6.16e-11	3.60e-08
CRT landfilling	Vinyl chloride	primary	5.74e-11	3.36e-08
US electric grid	Dichloromethane	model/secondary	5.46e-11	3.19e-08
CRT landfilling	Dichloromethane	primary	4.60e-11	2.69e-08
US electric grid	Cumene	model/secondary	3.62e-11	2.12e-08
CRT Incineration	Carbon tetrachloride	secondary	3.08e-11	1.80e-08
CRT landfilling	1,2-Dichloroethane	primary	2.87e-11	1.68e-08
CRT landfilling	Carbon tetrachloride	primary	2.87e-11	1.68e-08
CRT landfilling	Chloroform	primary	2.87e-11	1.68e-08
CRT landfilling	Tetrachloroethylene	primary	2.87e-11	1.68e-08
LPG Production	Isophorone	secondary	2.55e-11	1.49e-08
US electric grid	Phenanthrene	model/secondary	1.91e-11	1.12e-08
LPG Production	Toluene	secondary	1.81e-11	1.06e-08
US electric grid	Vinyl acetate	model/secondary	1.52e-11	8.89e-09
LPG Production	Acetaldehyde	secondary	1.46e-11	8.53e-09
LPG Production	Propionaldehyde	secondary	1.32e-11	7.74e-09
US electric grid	Biphenyl	model/secondary	1.16e-11	6.79e-09
US electric grid	Chloroform	model/secondary	1.11e-11	6.50e-09
LPG Production	Methyl ethyl ketone	secondary	1.07e-11	6.23e-09
US electric grid	Tetrachloroethylene	model/secondary	8.10e-12	4.74e-09
US electric grid	Ethyl Chloride	model/secondary	7.91e-12	4.63e-09
US electric grid	1,2-Dichloroethane	model/secondary	7.54e-12	4.41e-09
LPG Production	Acrolein	secondary	7.42e-12	4.34e-09
LPG Production	o-xylene	secondary	6.78e-12	3.96e-09
US electric grid	Fluorene	model/secondary	6.39e-12	3.74e-09
LPG Production	n-Propane	secondary	6.05e-12	3.54e-09
US electric grid	Fluoranthene	model/secondary	5.11e-12	2.99e-09
US electric grid	o-xylene	model/secondary	4.35e-12	2.54e-09
US electric grid	Acenaphthene	model/secondary	4.32e-12	2.53e-09
US electric grid	Ethylene dibromide	model/secondary	4.29e-12	2.51e-09
LPG Production	Ethylbenzene	secondary	4.17e-12	2.44e-09
US electric grid	Chlorobenzene	model/secondary	4.15e-12	2.42e-09
US electric grid	1,1,1-Trichloroethane	model/secondary	4.03e-12	2.36e-09
LPG Production	Bromomethane	secondary	3.68e-12	2.15e-09
US electric grid	Pyrene	model/secondary	2.54e-12	1.48e-09
LPG Production	Naphthalene	secondary	2.29e-12	1.34e-09
US electric grid	2,4-Dinitrotoluene	model/secondary	1.91e-12	1.12e-09
US electric grid	Acenaphthylene	model/secondary	1.72e-12	1.00e-09
US electric grid	Anthracene	model/secondary	1.48e-12	8.67e-10
US electric grid	2-Chloroacetophenone	model/secondary	1.32e-12	7.71e-10
LPG Production	Styrene	secondary	1.10e-12	6.42e-10
LPG Production	Benzyl chloride	secondary	8.49e-13	4.96e-10
US electric grid	Benzo[b,j,k]fluoranthene	model/secondary	8.10e-13	4.74e-10
LPG Production	Methyl tert-butyl ether	secondary	8.05e-13	4.70e-10
US electric grid	Chrysene	model/secondary	7.78e-13	4.55e-10
US electric grid	Benzo[a]anthracene	model/secondary	7.06e-13	4.13e-10
LPG Production	Phenol	secondary	7.03e-13	4.11e-10
LPG Production	Acetophenone	secondary	6.59e-13	3.85e-10
LPG Production	Methyl chloride	secondary	6.43e-13	3.76e-10

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
US electric grid	Indeno(1,2,3-cd)pyrene	model/secondary	5.02e-13	2.93e-10
LPG Production	Dichloromethane	secondary	3.52e-13	2.06e-10
LPG Production	Aromatic hydrocarbons	secondary	3.45e-13	2.02e-10
US electric grid	Benzo[g,h,i]perylene	model/secondary	2.74e-13	1.60e-10
US electric grid	Benzo[a]pyrene	model/secondary	2.59e-13	1.52e-10
LPG Production	Cumene	secondary	2.33e-13	1.36e-10
US electric grid	2-Methylnaphthalene	model/secondary	2.03e-13	1.19e-10
US electric grid	5-Methyl chrysene	model/secondary	1.50e-13	8.78e-11
LPG Production	Phenanthrene	secondary	1.49e-13	8.70e-11
LPG Production	Vinyl acetate	secondary	9.79e-14	5.72e-11
LPG Production	Biphenyl	secondary	7.47e-14	4.37e-11
LPG Production	Chloroform	secondary	7.16e-14	4.18e-11
LPG Production	1,4-Dichlorobenzene	secondary	5.44e-14	3.18e-11
LPG Production	Tetrachloroethylene	secondary	5.22e-14	3.05e-11
LPG Production	Ethyl Chloride	secondary	5.09e-14	2.98e-11
LPG Production	1,2-Dichloroethane	secondary	4.85e-14	2.84e-11
LPG Production	Fluorene	secondary	4.56e-14	2.66e-11
LPG Production	2-Methylnaphthalene	secondary	3.94e-14	2.31e-11
LPG Production	Fluoranthene	secondary	3.72e-14	2.17e-11
LPG Production	Acenaphthene	secondary	3.44e-14	2.01e-11
LPG Production	Ethylene dibromide	secondary	2.76e-14	1.61e-11
LPG Production	Chlorobenzene	secondary	2.67e-14	1.56e-11
LPG Production	1,1,1-Trichloroethane	secondary	2.43e-14	1.42e-11
LPG Production	Pyrene	secondary	2.36e-14	1.38e-11
LPG Production	Acenaphthylene	secondary	1.41e-14	8.22e-12
LPG Production	Anthracene	secondary	1.37e-14	8.01e-12
LPG Production	2,4-Dinitrotoluene	secondary	1.23e-14	7.20e-12
LPG Production	2-Chloroacetophenone	secondary	8.49e-15	4.96e-12
LPG Production	Benzo[a]anthracene	secondary	8.20e-15	4.79e-12
LPG Production	Chrysene	secondary	7.87e-15	4.60e-12
LPG Production	Indeno(1,2,3-cd)pyrene	secondary	6.10e-15	3.57e-12
LPG Production	Benzo[b,j,k]fluoranthene	secondary	4.83e-15	2.83e-12
LPG Production	Benzo[a]pyrene	secondary	4.49e-15	2.62e-12
LPG Production	Benzo[g,h,i]perylene	secondary	3.65e-15	2.13e-12
LPG Production	5-Methyl chrysene	secondary	9.67e-16	5.65e-13
LPG Production	Polycyclic aromatic hydrocarbons	secondary	3.83e-16	2.24e-13
Natural Gas Prod.	Polycyclic aromatic hydrocarbons	secondary	-4.51e-16	-2.63e-13
CRT Incineration	Polycyclic aromatic hydrocarbons	secondary	-6.98e-15	-4.08e-12
Natural Gas Prod.	5-Methyl chrysene	secondary	-9.74e-14	-5.70e-11
Fuel Oil #4 Prod.	Polycyclic aromatic hydrocarbons	secondary	-1.09e-13	-6.36e-11
Fuel Oil #4 Prod.	5-Methyl chrysene	secondary	-4.03e-13	-2.36e-10
Natural Gas Prod.	Aromatic hydrocarbons	secondary	-4.05e-13	-2.37e-10
Natural Gas Prod.	Benzo[b,j,k]fluoranthene	secondary	-4.87e-13	-2.85e-10
Natural Gas Prod.	2-Chloroacetophenone	secondary	-8.56e-13	-5.00e-10
Natural Gas Prod.	Benzo[a]pyrene	secondary	-1.20e-12	-7.02e-10
Natural Gas Prod.	Benzo[g,h,i]perylene	secondary	-1.22e-12	-7.12e-10
Natural Gas Prod.	2,4-Dinitrotoluene	secondary	-1.24e-12	-7.25e-10
Fuel Oil #4 Prod.	Benzo[g,h,i]perylene	secondary	-1.32e-12	-7.74e-10
Fuel Oil #4 Prod.	Benzo[a]pyrene	secondary	-1.79e-12	-1.05e-09

 Table M-17. CRT LCIA Results for the Photochemical Smog Impact Category

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Natural Gas Prod.	Indeno(1,2,3-cd)pyrene	secondary	-1.88e-12	-1.10e-09
Fuel Oil #4 Prod.	Benzo[b,j,k]fluoranthene	secondary	-2.01e-12	-1.18e-09
Natural Gas Prod.	Chrysene	secondary	-2.06e-12	-1.21e-09
Natural Gas Prod.	Benzo[a]anthracene	secondary	-2.14e-12	-1.25e-09
Fuel Oil #4 Prod.	Indeno(1,2,3-cd)pyrene	secondary	-2.28e-12	-1.34e-09
Natural Gas Prod.	1,1,1-Trichloroethane	secondary	-2.44e-12	-1.43e-09
Natural Gas Prod.	Acenaphthylene	secondary	-2.67e-12	-1.56e-09
Natural Gas Prod.	Chlorobenzene	secondary	-2.69e-12	-1.57e-09
Natural Gas Prod.	Ethylene dibromide	secondary	-2.78e-12	-1.63e-09
Fuel Oil #4 Prod.	Benzo[a]anthracene	secondary	-2.99e-12	-1.75e-09
Fuel Oil #4 Prod.	Chrysene	secondary	-3.01e-12	-1.76e-09
Natural Gas Prod.	Anthracene	secondary	-3.07e-12	-1.79e-09
Fuel Oil #4 Prod.	2-Chloroacetophenone	secondary	-3.54e-12	-2.07e-09
Natural Gas Prod.	1,2-Dichloroethane	secondary	-4.89e-12	-2.86e-09
Natural Gas Prod.	Acenaphthene	secondary	-5.05e-12	-2.95e-09
Fuel Oil #4 Prod.	2,4-Dinitrotoluene	secondary	-5.13e-12	-3.00e-09
Natural Gas Prod.	Ethyl Chloride	secondary	-5.13e-12	-3.00e-09
Natural Gas Prod.	Tetrachloroethylene	secondary	-5.26e-12	-3.07e-09
Fuel Oil #4 Prod.	Anthracene	secondary	-5.38e-12	-3.14e-09
Fuel Oil #4 Prod.	Acenaphthylene	secondary	-5.64e-12	-3.30e-09
Natural Gas Prod.	Fluoranthene	secondary	-5.87e-12	-3.43e-09
Natural Gas Prod.	Pyrene	secondary	-5.89e-12	-3.44e-09
CRT Incineration	Aromatic hydrocarbons	secondary	-6.29e-12	-3.68e-09
Natural Gas Prod.	Fluorene	secondary	-6.57e-12	-3.84e-09
CRT Incineration	2-Methylnaphthalene	secondary	-6.63e-12	-3.88e-09
Natural Gas Prod.	Chloroform	secondary	-7.21e-12	-4.22e-09
Natural Gas Prod.	Biphenyl	secondary	-7.53e-12	-4.40e-09
CRT Incineration	Dichlorobenzene (mixed isomers)	secondary	-9.15e-12	-5.35e-09
Fuel Oil #4 Prod.	Pyrene	secondary	-9.18e-12	-5.37e-09
Natural Gas Prod.	Vinyl acetate	secondary	-9.86e-12	-5.77e-09
Fuel Oil #4 Prod.	1,1,1-Trichloroethane	secondary	-1.01e-11	-5.91e-09
Fuel Oil #4 Prod.	Chlorobenzene	secondary	-1.11e-11	-6.50e-09
Fuel Oil #4 Prod.	Ethylene dibromide	secondary	-1.15e-11	-6.72e-09
Fuel Oil #4 Prod.	Acenaphthene	secondary	-1.30e-11	-7.58e-09
Fuel Oil #4 Prod.	2-Methylnaphthalene	secondary	-1.38e-11	-8.07e-09
Fuel Oil #4 Prod.	Fluoranthene	secondary	-1.50e-11	-8.78e-09
Fuel Oil #4 Prod.	Fluorene	secondary	-1.85e-11	-1.08e-08
Fuel Oil #4 Prod.	1,4-Dichlorobenzene	secondary	-1.91e-11	-1.11e-08
Fuel Oil #4 Prod.	1,2-Dichloroethane	secondary	-2.02e-11	-1.18e-08
Natural Gas Prod.	2-Methylnaphthalene	secondary	-2.06e-11	-1.21e-08
Fuel Oil #4 Prod.	Ethyl Chloride	secondary	-2.12e-11	-1.24e-08
Fuel Oil #4 Prod.	Tetrachloroethylene	secondary	-2.17e-11	-1.27e-08
Natural Gas Prod.	Cumene	secondary	-2.35e-11	-1.37e-08
CRT Incineration	5-Methyl chrysene	secondary	-2.49e-11	-1.46e-08
Natural Gas Prod.	Phenanthrene	secondary	-2.69e-11	-1.57e-08
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	-2.85e-11	-1.67e-08
Fuel Oil #4 Prod.	Chloroform	secondary	-2.98e-11	-1.74e-08
CRT Incineration	Benzo[g,h,i]perylene	secondary	-3.09e-11	-1.81e-08
Fuel Oil #4 Prod.	Biphenyl	secondary	-3.11e-11	-1.82e-08

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Natural Gas Prod.	Dichloromethane	secondary	-3.54e-11	-2.07e-08
Fuel Oil #4 Prod.	Vinyl acetate	secondary	-4.08e-11	-2.38e-08
CRT Incineration	Benzo[a]pyrene	secondary	-4.34e-11	-2.53e-08
Fuel Oil #4 Prod.	Phenanthrene	secondary	-5.99e-11	-3.50e-08
Natural Gas Prod.	Methyl chloride	secondary	-6.48e-11	-3.79e-08
Natural Gas Prod.	Acetophenone	secondary	-6.64e-11	-3.88e-08
CRT Incineration	Indeno(1,2,3-cd)pyrene	secondary	-6.95e-11	-4.07e-08
Natural Gas Prod.	Phenol	secondary	-7.09e-11	-4.14e-08
Natural Gas Prod.	Methyl tert-butyl ether	secondary	-8.11e-11	-4.74e-08
Natural Gas Prod.	Benzyl chloride	secondary	-8.56e-11	-5.00e-08
CRT Incineration	Benzo[a]anthracene	secondary	-9.11e-11	-5.32e-08
Fuel Oil #4 Prod.	Cumene	secondary	-9.70e-11	-5.67e-08
Fuel Oil #4 Prod.	Aromatic hydrocarbons	secondary	-9.80e-11	-5.73e-08
Natural Gas Prod.	Styrene	secondary	-1.11e-10	-6.47e-08
CRT Incineration	Chrysene	secondary	-1.14e-10	-6.65e-08
CRT Incineration	Benzo[b,j,k]fluoranthene	secondary	-1.24e-10	-7.28e-08
Fuel Oil #4 Prod.	Dichloromethane	secondary	-1.47e-10	-8.57e-08
CRT Incineration	Anthracene	secondary	-2.38e-10	-1.39e-07
Fuel Oil #4 Prod.	Methyl chloride	secondary	-2.68e-10	-1.57e-07
Fuel Oil #4 Prod.	Acetophenone	secondary	-2.75e-10	-1.61e-07
CRT Incineration	Acenaphthylene	secondary	-2.83e-10	-1.66e-07
Fuel Oil #4 Prod.	Phenol	secondary	-2.93e-10	-1.71e-07
CRT Incineration	2,4-Dinitrotoluene	secondary	-3.17e-10	-1.85e-07
Fuel Oil #4 Prod.	Methyl tert-butyl ether	secondary	-3.35e-10	-1.96e-07
Fuel Oil #4 Prod.	Benzyl chloride	secondary	-3.54e-10	-2.07e-07
Natural Gas Prod.	Bromomethane	secondary	-3.71e-10	-2.17e-07
CRT Incineration	Pyrene	secondary	-3.75e-10	-2.19e-07
Natural Gas Prod.	Ethylbenzene	secondary	-4.21e-10	-2.46e-07
Fuel Oil #4 Prod.	Styrene	secondary	-4.58e-10	-2.68e-07
CRT Incineration	Acenaphthene	secondary	-5.78e-10	-3.38e-07
CRT Incineration	1,1,1-Trichloroethane	secondary	-6.24e-10	-3.65e-07
Natural Gas Prod.	n-Propane	secondary	-6.72e-10	-3.93e-07
Natural Gas Prod.	Naphthalene	secondary	-6.72e-10	-3.93e-07
CRT Incineration	Chlorobenzene	secondary	-6.87e-10	-4.02e-07
CRT Incineration	Ethylene dibromide	secondary	-7.10e-10	-4.15e-07
Natural Gas Prod.	Acrolein	secondary	-7.48e-10	-4.37e-07
CRT Incineration	Fluoranthene	secondary	-8.04e-10	-4.70e-07
Fuel Oil #4 Prod.	Naphthalene	secondary	-8.25e-10	-4.82e-07
CRT Incineration	Fluorene	secondary	-1.03e-09	-6.02e-07
Natural Gas Prod.	Methyl ethyl ketone	secondary	-1.07e-09	-6.28e-07
CRT Incineration	1,2-Dichloroethane	secondary	-1.22e-09	-7.12e-07
CRT Incineration	Ethyl Chloride	secondary	-1.31e-09	-7.67e-07
CRT Incineration	Tetrachloroethylene	secondary	-1.31e-09	-7.67e-07
Natural Gas Prod.	Propionaldehyde	secondary	-1.33e-09	-7.80e-07
Natural Gas Prod.	Acetaldehyde	secondary	-1.47e-09	-8.59e-07
Fuel Oil #4 Prod.	Bromomethane	secondary	-1.53e-09	-8.96e-07
Fuel Oil #4 Prod.	Ethylbenzene	secondary	-1.73e-09	-1.01e-06
CRT Incineration	Chloroform	secondary	-1.81e-09	-1.06e-06
CRT Incineration	Biphenyl	secondary	-1.92e-09	-1.12e-06

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Fuel Oil #4 Prod.	o-xylene	secondary	-2.47e-09	-1.45e-06
Fuel Oil #4 Prod.	n-Propane	secondary	-2.51e-09	-1.47e-06
CRT Incineration	Vinyl acetate	secondary	-2.52e-09	-1.47e-06
Natural Gas Prod.	Isophorone	secondary	-2.57e-09	-1.50e-06
CRT Incineration	Phenanthrene	secondary	-3.06e-09	-1.79e-06
Fuel Oil #4 Prod.	Acrolein	secondary	-3.09e-09	-1.81e-06
Fuel Oil #4 Prod.	Methyl ethyl ketone	secondary	-4.44e-09	-2.60e-06
Fuel Oil #4 Prod.	Propionaldehyde	secondary	-5.51e-09	-3.22e-06
CRT Incineration	Cumene	secondary	-6.00e-09	-3.51e-06
Fuel Oil #4 Prod.	Acetaldehyde	secondary	-6.08e-09	-3.55e-06
Fuel Oil #4 Prod.	Toluene	secondary	-6.85e-09	-4.00e-06
CRT Incineration	Dichloromethane	secondary	-9.01e-09	-5.27e-06
Fuel Oil #4 Prod.	Isophorone	secondary	-1.06e-08	-6.21e-06
Natural Gas Prod.	Toluene	secondary	-1.46e-08	-8.52e-06
CRT Incineration	Methyl chloride	secondary	-1.65e-08	-9.67e-06
CRT Incineration	Acetophenone	secondary	-1.70e-08	-9.92e-06
CRT Incineration	Phenol	secondary	-1.81e-08	-1.06e-05
CRT Incineration	Methyl tert-butyl ether	secondary	-2.07e-08	-1.21e-05
CRT Incineration	Naphthalene	secondary	-2.09e-08	-1.22e-05
CRT Incineration	Benzyl chloride	secondary	-2.19e-08	-1.28e-05
CRT Incineration	Xvlene (mixed isomers)	secondary	-2.44e-08	-1.43e-05
Natural Gas Prod.	o-xvlene	secondary	-2.64e-08	-1.54e-05
CRT Incineration	Styrene	secondary	-2.83e-08	-1.65e-05
Natural Gas Prod.	Formaldehvde	secondary	-5.11e-08	-2.99e-05
CRT Incineration	Bromomethane	secondary	-9.47e-08	-5.54e-05
CRT Incineration	Ethylbenzene	secondary	-1.03e-07	-6.00e-05
CRT Incineration	n-Propane	secondary	-1 52e-07	-8 90e-05
CRT Incineration	Acrolein	secondary	-1.91e-07	-1.12e-04
CRT Incineration	Toluene	secondary	-2.25e-07	-1.31e-04
CRT Incineration	Methyl ethyl ketone	secondary	-2.74e-07	-1 60e-04
CRT Incineration	Butane	secondary	-3 03e-07	-1 77e-04
CRT Incineration	Hexane	secondary	-3 13e-07	-1 83e-04
CRT Incineration	Pronionaldehyde	secondary	-3 41e-07	-1 99e-04
CRT Incineration	Acetaldehyde	secondary	-3 75e-07	-2 20e-04
CRT Incineration	Pentane	secondary	-3 85e-07	-2 25e-04
CRT Incineration	Fthane	secondary	-4 48e-07	-2 62e-04
Fuel Oil #4 Prod	Hexane	secondary	-5 67e-07	-3 31e-04
Fuel Oil #4 Prod	Butane	secondary	-6 32e-07	-3 69e-04
CRT Incineration	Isophorone	secondary	-6 56e-07	-3 84e-04
Fuel Oil #4 Prod	Pentane	secondary	-8 02e-07	-4 69e-04
Natural Gas Prod	Heyane	secondary	-8 46e-07	-4.95e-04
Fuel Oil #4 Prod	Ethane	secondary	-9.33e-07	-5.45e-04
Natural Gas Prod	Butane	secondary	-9.35e-07	-5.52e-04
Natural Gas Prod	Pentane	secondary	-1 20e-06	-7.01e-04
Natural Gas Prod	Fthane	secondary	-1.200-00	-8 15e-04
CRT Incineration	Formaldehyde	secondary	-1.590-00	-9.60-04
Fuel Oil #4 Prod	Aldehydes	secondary	-1.040-00	-1.17-02
Fuel Oil #4 Prod	Formaldehyde	secondary	-2.000-00	-1.1/0-03
Natural Gas Prod	Aldebydes	secondary	-2.030-00	1 400 02
Ivatural Gas FIGU.	Auchydes	secondary	-2.396-00	-1.406-03

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene-	% of Total
			equivalents)	
Fuel Oil #4 Prod.	Benzene	secondary	-8.19e-06	-4.79e-03
CRT Incineration	Benzene	secondary	-1.00e-05	-5.87e-03
Natural Gas Prod.	Hydrocarbons, remaining unspeciated	secondary	-1.24e-05	-7.26e-03
Fuel Oil #4 Prod.	Methane	secondary	-1.96e-05	-1.15e-02
CRT Incineration	Aldehydes	secondary	-6.76e-05	-3.95e-02
CRT Incineration	Methane	secondary	-1.06e-04	-6.20e-02
CRT Incineration	Hydrocarbons, remaining unspeciated	secondary	-1.07e-04	-6.27e-02
Natural Gas Prod.	Benzene	secondary	-1.20e-04	-7.02e-02
CRT Incineration	Nonmethane hydrocarbons, remaining unspeciated	secondary	-1.54e-04	-9.02e-02
Fuel Oil #4 Prod.	Nonmethane hydrocarbons, remaining unspeciated	secondary	-1.56e-04	-9.11e-02
Natural Gas Prod.	Methane	secondary	-1.76e-04	-1.03e-01
Fuel Oil #4 Prod.	Hydrocarbons, remaining unspeciated	secondary	-2.32e-04	-1.36e-01
Natural Gas Prod.	Nonmethane hydrocarbons, remaining unspeciated	secondary	-4.52e-04	-2.64e-01
Total End-of-life			-1.53e-03	-8.95e-01
Total All Life-cycle Stages			1.71e-01	1.00e+02

 Table M-17. CRT LCIA Results for the Photochemical Smog Impact Category

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Materials Processing Life-cycle	Stage		,	
Natural Gas Prod.	Nonmethane hydrocarbons, remaining unspeciated	secondary	6.30e-02	4.45e+01
Natural Gas Prod.	Methane	secondary	2.45e-02	1.73e+01
Natural Gas Prod.	Benzene	secondary	1.67e-02	1.18e+01
Steel Prod., cold-rolled, semi-finished	Nonmethane hydrocarbons, remaining unspeciated	secondary	1.62e-02	1.15e+01
Natural Gas Prod.	Hydrocarbons, remaining unspeciated	secondary	1.73e-03	1.22e+00
Steel Prod., cold-rolled, semi-finished	Hydrocarbons, remaining unspeciated	secondary	1.68e-03	1.19e+00
PMMA Sheet Prod.	Nonmethane hydrocarbons, remaining unspeciated	secondary	1.24e-03	8.74e-01
Polycarbonate Production	Nonmethane hydrocarbons, remaining unspeciated	secondary	1.03e-03	7.26e-01
Aluminum Prod.	Nonmethane hydrocarbons, remaining unspeciated	secondary	5.48e-04	3.87e-01
PET Resin Production	Nonmethane hydrocarbons, remaining unspeciated	secondary	5.06e-04	3.58e-01
Natural Gas Prod.	Aldehvdes	secondary	3.33e-04	2.36e-01
Styrene-butadiene Copolymer Prod.	Hydrocarbons, remaining unspeciated	secondary	2.88e-04	2.04e-01
Natural Gas Prod.	Ethane	secondary	1.94e-04	1.37e-01
Natural Gas Prod.	Pentane	secondary	1.67e-04	1.18e-01
Natural Gas Prod.	Butane	secondary	1.32e-04	9.30e-02
Natural Gas Prod.	Hexane	secondary	1.18e-04	8.34e-02
Steel Prod., cold-rolled, semi-finished	Ethylene	secondary	1.02e-04	7.18e-02
PMMA Sheet Prod.	Methane	secondary	8.05e-05	5.69e-02
Polycarbonate Production	Methane	secondary	7.94e-05	5.62e-02
Steel Prod., cold-rolled, semi-finished	Ethane	secondary	4.80e-05	3.39e-02
Steel Prod., cold-rolled, semi-finished	Methane	secondary	3.74e-05	2.64e-02
Polycarbonate Production	Aromatic hydrocarbons	secondary	3.49e-05	2.47e-02
Styrene-butadiene Copolymer Prod.	Aromatic hydrocarbons	secondary	2.26e-05	1.60e-02
Styrene-butadiene Copolymer Prod.	Methane	secondary	2.10e-05	1.49e-02
Aluminum Prod.	Methane	secondary	2.10e-05	1.48e-02
Steel Prod., cold-rolled, semi-finished	n-Propane	secondary	1.88e-05	1.33e-02
PMMA Sheet Prod.	Aldehvdes	secondary	1.87e-05	1.32e-02
Polycarbonate Production	Aldehydes	secondary	1.46e-05	1.03e-02
Steel Prod., cold-rolled, semi-finished	Propylene	secondary	1.22e-05	8.62e-03
Steel Prod., cold-rolled, semi-finished	Aldehvdes	secondary	9.25e-06	6.54e-03
Natural Gas Prod.	Formaldehyde	secondary	7.11e-06	5.03e-03
PET Resin Production	Methane	secondary	6.99e-06	4.94e-03
Aluminum Prod.	Xylene (mixed isomers)	secondary	5.24e-06	3.71e-03
Aluminum Prod.	Polycyclic aromatic hydrocarbons	secondary	5.13e-06	3.63e-03
Steel Prod., cold-rolled, semi-finished	Butane	secondary	4.43e-06	3.13e-03
Natural Gas Prod.	o-xylene	secondary	3.68e-06	2.60e-03
Steel Prod., cold-rolled, semi-finished	Pentane	secondary	3.39e-06	2.40e-03
Steel Prod., cold-rolled, semi-finished	Benzene	secondary	2.58e-06	1.82e-03
Steel Prod., cold-rolled, semi-finished	Toluene	secondary	2.14e-06	1.51e-03
Steel Prod., cold-rolled, semi-finished	Xylene (mixed isomers)	secondary	2.04e-06	1.44e-03
Natural Gas Prod.	Toluene	secondary	2.03e-06	1.44e-03
Steel Prod., cold-rolled, semi-finished	Acetylene	secondary	1.92e-06	1.36e-03
Aluminum Prod.	Toluene	secondary	1.74e-06	1.23e-03
Aluminum Prod.	Aldehydes	secondary	1.71e-06	1.21e-03
Steel Prod., cold-rolled, semi-finished	Hexane	secondary	9.03e-07	6.39e-04
Steel Prod., cold-rolled, semi-finished	Formaldehyde	secondary	7.47e-07	5.28e-04
Steel Prod., cold-rolled, semi-finished	Phenol	secondary	7.00e-07	4.95e-04
PMMA Sheet Prod.	Aromatic hydrocarbons	secondary	5.84e-07	4.13e-04
Aluminum Prod.	Alcohols	secondary	5.58e-07	3.94e-04

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Aluminum Prod.	Benzene	secondary	4.55e-07	3.21e-04
Steel Prod., cold-rolled, semi-finished	Heptane	secondary	4.32e-07	3.05e-04
Natural Gas Prod.	Isophorone	secondary	3.58e-07	2.53e-04
Steel Prod., cold-rolled, semi-finished	Aromatic hydrocarbons	secondary	2.47e-07	1.75e-04
Steel Prod., cold-rolled, semi-finished	Methanol	secondary	2.17e-07	1.53e-04
Natural Gas Prod.	Acetaldehyde	secondary	2.05e-07	1.45e-04
Natural Gas Prod.	Propionaldehyde	secondary	1.86e-07	1.31e-04
Steel Prod., cold-rolled, semi-finished	Ethanol	secondary	1.74e-07	1.23e-04
Natural Gas Prod.	Methyl ethyl ketone	secondary	1.50e-07	1.06e-04
Steel Prod., cold-rolled, semi-finished	Acetaldehyde	secondary	1.45e-07	1.03e-04
Steel Prod., cold-rolled, semi-finished	Ethylbenzene	secondary	1.15e-07	8.14e-05
Natural Gas Prod.	Acrolein	secondary	1.04e-07	7.36e-05
Polycarbonate Production	Ethanethiol	secondary	1.03e-07	7.26e-05
Natural Gas Prod.	Naphthalene	secondary	9.37e-08	6.62e-05
Natural Gas Prod.	n-Propane	secondary	9.36e-08	6.61e-05
PMMA Sheet Prod.	Ethanethiol	secondary	7.63e-08	5.40e-05
Natural Gas Prod.	Ethylbenzene	secondary	5.87e-08	4.15e-05
Steel Prod., cold-rolled, semi-finished	Acetone	secondary	5.80e-08	4.10e-05
Natural Gas Prod	Bromomethane	secondary	5 16e-08	3 65e-05
PET Resin Production	Aldehydes	secondary	4 02e-08	2 84e-05
Styrene-butadiene Copolymer Prod	Aldehydes	secondary	4 01e-08	2.81e-05
Steel Prod cold-rolled semi-finished	Benzo[a]nvrene	secondary	3 33e-08	2.35e-05
Steel Prod. cold-rolled semi-finished	Polycyclic aromatic hydrocarbons	secondary	1.82e-08	1.29e-05
Natural Gas Prod	Styrene	secondary	1.54e-08	1.09e-05
Natural Gas Prod	Benzyl chloride	secondary	1.540 08	8.43e-06
Natural Gas Prod	Methyl tert butyl ether	secondary	1.13e.08	7.98e.06
Natural Gas Prod	Phenol	secondary	9.87e-00	6.98e-06
Natural Gas Prod	Acetophenone	secondary	9.25e.00	6.54e.06
Natural Gas Prod	Methyl chloride	secondary	9.02e-09	6.38e-06
Steel Bred cold rolled somi finished	Nanhthalana	secondary	5.550.00	2.020.06
Natural Gas Brad	Dishlaromathana	secondary	1.04o.09	3.920-00
Natural Gas Prod.	1 4 Dishlarahangana	secondary	4.946-09	2.800.06
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	3.976-09	2.808-00
Natural Gas Prod.	Comment	secondary	3.730-09	2.03e-00
Natural Gas Prod.	2 Mathada and thalan a	secondary	3.276-09	2.31e-00
Natural Gas Prod.	2-Methyinaphthalene	secondary	2.886-09	2.03e-06
Natural Cas Dual	Visual a state	secondary	1.876-09	0.71-07
Natural Gas Prod.	v inyl acetate	secondary	1.376-09	9.71e-07
Aluminum Prod.	Phenol	secondary	1.13e-09	8.00e-07
Natural Gas Prod.	Bipnenyi	secondary	1.05e-09	7.41e-07
Natural Gas Prod.	Chloroform	secondary	1.00e-09	7.10e-07
Natural Gas Prod.	Fluorene	secondary	9.15e-10	6.47e-07
Natural Gas Prod.	Pyrene	secondary	8.20e-10	5.80e-07
Natural Gas Prod.	Fluoranthene	secondary	8.17e-10	5.78e-07
Natural Gas Prod.	Tetrachloroethylene	secondary	7.32e-10	5.18e-07
Natural Gas Prod.	Ethyl Chloride	secondary	7.15e-10	5.06e-07
Natural Gas Prod.	Acenaphthene	secondary	7.04e-10	4.98e-07
Natural Gas Prod.	1,2-Dichloroethane	secondary	6.81e-10	4.81e-07
Natural Gas Prod.	Anthracene	secondary	4.27e-10	3.02e-07

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Natural Gas Prod.	Ethylene dibromide	secondary	3.87e-10	2.74e-07
Natural Gas Prod.	Chlorobenzene	secondary	3.75e-10	2.65e-07
Natural Gas Prod.	Acenaphthylene	secondary	3.72e-10	2.63e-07
Natural Gas Prod.	1,1,1-Trichloroethane	secondary	3.40e-10	2.41e-07
Natural Gas Prod.	Benzo[a]anthracene	secondary	2.98e-10	2.11e-07
Natural Gas Prod.	Chrysene	secondary	2.87e-10	2.03e-07
Natural Gas Prod.	Indeno(1,2,3-cd)pyrene	secondary	2.62e-10	1.85e-07
Natural Gas Prod.	2,4-Dinitrotoluene	secondary	1.73e-10	1.22e-07
Natural Gas Prod.	Benzo[g,h,i]perylene	secondary	1.70e-10	1.20e-07
Natural Gas Prod.	Benzo[a]pyrene	secondary	1.67e-10	1.18e-07
Natural Gas Prod.	2-Chloroacetophenone	secondary	1.19e-10	8.43e-08
Natural Gas Prod.	Benzo[b,j,k]fluoranthene	secondary	6.79e-11	4.80e-08
Natural Gas Prod.	Aromatic hydrocarbons	secondary	5.64e-11	3.99e-08
Natural Gas Prod.	5-Methyl chrysene	secondary	1.36e-11	9.60e-09
Natural Gas Prod.	Polycyclic aromatic hydrocarbons	secondary	6.28e-14	4.44e-11
Steel Prod., cold-rolled, semi-finished	Propionaldehyde	secondary	4.58e-14	3.23e-11
Steel Prod., cold-rolled, semi-finished	Benzaldehyde	secondary	1.23e-14	8.71e-12
Total Materials Processing			1.29e-01	9.12e+01
Manufacturing Life-cycle Stage	9			
Monitor/module	Isopropyl alcohol	primary	3.48e-03	2.46e+00
LPG Production	Hydrocarbons, remaining unspeciated	secondary	2.99e-03	2.11e+00
LPG Production	Nonmethane hydrocarbons, remaining unspeciated	secondary	2.04e-03	1.44e+00
Natural Gas Prod.	Nonmethane hydrocarbons, remaining unspeciated	secondary	1.42e-03	1.00e+00
Natural Gas Prod.	Methane	secondary	5.54e-04	3.91e-01
Natural Gas Prod.	Benzene	secondary	3.77e-04	2.67e-01
LPG Production	Methane	secondary	2.82e-04	1.99e-01
LPG Production	Benzene	secondary	1.30e-04	9.20e-02
Natural Gas Prod.	Hydrocarbons, remaining unspeciated	secondary	3.91e-05	2.76e-02
Backlight	Diethyl ether	primary	3.69e-05	2.61e-02
Fuel Oil #4 Prod.	Hydrocarbons, remaining unspeciated	secondary	3.32e-05	2.35e-02
Monitor/module	Acetone	primary	3.31e-05	2.34e-02
LPG Production	Aldehydes	secondary	3.17e-05	2.24e-02
Panel components	Heptane	primary	3.09e-05	2.19e-02
Panel components	Nonmethane hydrocarbons, remaining unspeciated	primary	3.09e-05	2.19e-02
Panel components	Toluene	primary	3.06e-05	2.17e-02
LPG Production	Formaldehyde	secondary	2.36e-05	1.67e-02
Fuel Oil #4 Prod.	Nonmethane hydrocarbons, remaining unspeciated	secondary	2.23e-05	1.58e-02
Monitor/module	Cyclohexane	primary	1.93e-05	1.36e-02
Fuel Oil #6 Prod.	Hydrocarbons, remaining unspeciated	secondary	1.83e-05	1.29e-02
LPG Production	Ethane	secondary	1.48e-05	1.05e-02
Japanese Electric Grid	Formaldehyde	model/secondary	1.44e-05	1.02e-02
LPG Production	Pentane	secondary	1.27e-05	9.00e-03
Backlight	Ethanol	primary	1.24e-05	8.76e-03
Fuel Oil #6 Prod.	Nonmethane hydrocarbons, remaining unspeciated	secondary	1.21e-05	8.55e-03
LPG Production	Butane	secondary	1.00e-05	7.09e-03
Fuel Oil #2 Prod.	Hydrocarbons, remaining unspeciated	secondary	9.21e-06	6.51e-03
LPG Production	Hexane	secondary	9.00e-06	6.36e-03
US electric grid	Methane	model/secondary	8.74e-06	6.18e-03
Natural Gas Prod.	Aldehydes	secondary	7.53e-06	5.32e-03

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
PWB Mfg.	Formaldehyde	model/secondary	7.21e-06	5.09e-03
Fuel Oil #2 Prod.	Nonmethane hydrocarbons, remaining unspeciated	secondary	6.26e-06	4.42e-03
Natural Gas Prod.	Ethane	secondary	4.38e-06	3.10e-03
Natural Gas Prod.	Pentane	secondary	3.77e-06	2.66e-03
Natural Gas Prod.	Butane	secondary	2.97e-06	2.10e-03
Panel components	HCFC-225ca	primary	2.94e-06	2.08e-03
Panel components	HCFC-225cb	primary	2.94e-06	2.08e-03
Fuel Oil #4 Prod.	Methane	secondary	2.81e-06	1.99e-03
Japanese Electric Grid	Toluene	model/secondary	2.71e-06	1.91e-03
Natural Gas Prod.	Hexane	secondary	2.66e-06	1.88e-03
Japanese Electric Grid	Methane	model/secondary	1.91e-06	1.35e-03
Japanese Electric Grid	Isophorone	model/secondary	1.70e-06	1.20e-03
Fuel Oil #6 Prod.	Methane	secondary	1.40e-06	9.92e-04
Fuel Oil #4 Prod.	Benzene	secondary	1.17e-06	8.30e-04
Japanese Electric Grid	Acetaldehyde	model/secondary	9.71e-07	6.86e-04
Japanese Electric Grid	Benzene	model/secondary	9.69e-07	6.85e-04
Japanese Electric Grid	Propionaldehyde	model/secondary	8.81e-07	6.23e-04
Fuel Oil #2 Prod.	Methane	secondary	8.39e-07	5.93e-04
Japanese Electric Grid	Methyl ethyl ketone	model/secondary	7.09e-07	5.01e-04
Japanese Electric Grid	Naphthalene	model/secondary	5.66e-07	4.00e-04
Fuel Oil #6 Prod.	Benzene	secondary	5.27e-07	3.73e-04
Japanese Electric Grid	Acrolein	model/secondary	4.94e-07	3.49e-04
Fuel Oil #2 Prod.	Benzene	secondary	3.76e-07	2.66e-04
Panel components	Methyl ethyl ketone	primary	3.46e-07	2.45e-04
Japanese Electric Grid	Ethylbenzene	model/secondary	3.04e-07	2.15e-04
Fuel Oil #4 Prod.	Formaldehyde	secondary	2.91e-07	2.06e-04
Fuel Oil #4 Prod.	Aldehydes	secondary	2.86e-07	2.02e-04
Japanese Electric Grid	Bromomethane	model/secondary	2.45e-07	1.73e-04
Fuel Oil #6 Prod.	Formaldehyde	secondary	1.71e-07	1.21e-04
Natural Gas Prod.	Formaldehyde	secondary	1.61e-07	1.13e-04
LPG Production	Isophorone	secondary	1.42e-07	1.00e-04
Fuel Oil #4 Prod.	Ethane	secondary	1.34e-07	9.45e-05
Fuel Oil #6 Prod.	Aldehydes	secondary	1.28e-07	9.07e-05
Fuel Oil #4 Prod.	Pentane	secondary	1.15e-07	8.13e-05
Japanese Electric Grid	Xylene (mixed isomers)	model/secondary	1.08e-07	7.65e-05
Japanese Electric Grid	Hexane	model/secondary	1.07e-07	7.57e-05
LPG Production	Toluene	secondary	1.00e-07	7.10e-05
Fuel Oil #2 Prod.	Aldehydes	secondary	9.17e-08	6.48e-05
Fuel Oil #4 Prod.	Butane	secondary	9.06e-08	6.40e-05
Natural Gas Prod.	o-xylene	secondary	8.30e-08	5.87e-05
Fuel Oil #4 Prod.	Hexane	secondary	8.12e-08	5.74e-05
LPG Production	Acetaldehyde	secondary	8.11e-08	5.73e-05
US electric grid	Isophorone	model/secondary	7.66e-08	5.42e-05
Fuel Oil #2 Prod.	Formaldehyde	secondary	7.56e-08	5.35e-05
LPG Production	Propionaldehyde	secondary	7.36e-08	5.20e-05
Japanese Electric Grid	Styrene	model/secondary	7.31e-08	5.17e-05
US electric grid	Formaldehyde	model/secondary	6.90e-08	4.88e-05
Fuel Oil #6 Prod.	Ethane	secondary	6.01e-08	4.25e-05
LPG Production	Methyl ethyl ketone	secondary	5.92e-08	4.19e-05

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Japanese Electric Grid	Benzyl chloride	model/secondary	5.65e-08	4.00e-05
Japanese Electric Grid	Methyl tert-butyl ether	model/secondary	5.34e-08	3.77e-05
Fuel Oil #6 Prod.	Pentane	secondary	5.17e-08	3.65e-05
Japanese Electric Grid	o-xylene	model/secondary	4.94e-08	3.49e-05
Japanese Electric Grid	Phenol	model/secondary	4.68e-08	3.31e-05
Natural Gas Prod.	Toluene	secondary	4.58e-08	3.24e-05
Japanese Electric Grid	Acetophenone	model/secondary	4.39e-08	3.10e-05
US electric grid	Acetaldehyde	model/secondary	4.38e-08	3.10e-05
Fuel Oil #2 Prod.	Ethane	secondary	4.28e-08	3.03e-05
Japanese Electric Grid	Methyl chloride	model/secondary	4.28e-08	3.03e-05
US electric grid	Benzene	model/secondary	4.27e-08	3.02e-05
LPG Production	Acrolein	secondary	4.12e-08	2.92e-05
Fuel Oil #6 Prod.	Butane	secondary	4.07e-08	2.88e-05
US electric grid	Propionaldehyde	model/secondary	3.98e-08	2.81e-05
LPG Production	o-xylene	secondary	3.77e-08	2.66e-05
Fuel Oil #2 Prod.	Pentane	secondary	3.68e-08	2.60e-05
Fuel Oil #6 Prod.	Hexane	secondary	3.65e-08	2.58e-05
LPG Production	n-Propane	secondary	3.36e-08	2.38e-05
US electric grid	Methyl ethyl ketone	model/secondary	3.20e-08	2.26e-05
Fuel Oil #2 Prod.	Butane	secondary	2.90e-08	2.05e-05
US electric grid	Toluene	model/secondary	2.77e-08	1.96e-05
Fuel Oil #2 Prod.	Hexane	secondary	2.60e-08	1.84e-05
Japanese Electric Grid	Dichloromethane	model/secondary	2.34e-08	1.66e-05
LPG Production	Ethylbenzene	secondary	2.32e-08	1.64e-05
US electric grid	Acrolein	model/secondary	2.23e-08	1.58e-05
LPG Production	Bromomethane	secondary	2.04e-08	1.45e-05
Japanese Electric Grid	Cumene hydroperoxide	model/secondary	1.55e-08	1.10e-05
Japanese Electric Grid	Phenanthrene	model/secondary	1.33e-08	9.42e-06
LPG Production	Naphthalene	secondary	1.27e-08	9.01e-06
US electric grid	Ethylbenzene	model/secondary	1.25e-08	8.82e-06
US electric grid	Bromomethane	model/secondary	1.11e-08	7.82e-06
Japanese Electric Grid	Acenaphthene	model/secondary	1.11e-08	7.81e-06
Natural Gas Prod.	Isophorone	secondary	8.08e-09	5.71e-06
Japanese Electric Grid	Vinyl acetate	model/secondary	6.50e-09	4.60e-06
LPG Production	Styrene	secondary	6.11e-09	4.32e-06
Japanese Electric Grid	Biphenyl	model/secondary	4.97e-09	3.52e-06
US electric grid	Xylene (mixed isomers)	model/secondary	4.89e-09	3.46e-06
US electric grid	Hexane	model/secondary	4.84e-09	3.42e-06
Japanese Electric Grid	Chloroform	model/secondary	4.76e-09	3.37e-06
LPG Production	Benzyl chloride	secondary	4.72e-09	3.34e-06
Japanese Electric Grid	Fluorene	model/secondary	4.68e-09	3.31e-06
Natural Gas Prod.	Acetaldehyde	secondary	4.62e-09	3.27e-06
Japanese Electric Grid	1,1,1-Trichloroethane	model/secondary	4.57e-09	3.23e-06
LPG Production	Methyl tert-butyl ether	secondary	4.47e-09	3.16e-06
Japanese Electric Grid	Fluoranthene	model/secondary	4.47e-09	3.16e-06
Natural Gas Prod.	Propionaldehyde	secondary	4.19e-09	2.96e-06
LPG Production	Phenol	secondary	3.91e-09	2.76e-06
LPG Production	Acetophenone	secondary	3.66e-09	2.59e-06
LPG Production	Methyl chloride	secondary	3.57e-09	2.53e-06
Japanese Electric Grid	Tetrachloroethylene	model/secondary	3.47e-09	2.45e-06

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Japanese Electric Grid	Ethyl Chloride	model/secondary	3.39e-09	2.40e-06
Natural Gas Prod.	Methyl ethyl ketone	secondary	3.38e-09	2.39e-06
US electric grid	Styrene	model/secondary	3.30e-09	2.34e-06
Japanese Electric Grid	1,2-Dichloroethane	model/secondary	3.23e-09	2.28e-06
Japanese Electric Grid	Pyrene	model/secondary	3.23e-09	2.28e-06
US electric grid	Naphthalene	model/secondary	2.69e-09	1.90e-06
US electric grid	Benzyl chloride	model/secondary	2.55e-09	1.80e-06
US electric grid	Methyl tert-butyl ether	model/secondary	2.42e-09	1.71e-06
Natural Gas Prod.	Acrolein	secondary	2.35e-09	1.66e-06
US electric grid	Phenol	model/secondary	2.11e-09	1.49e-06
Natural Gas Prod.	Naphthalene	secondary	2.11e-09	1.49e-06
Natural Gas Prod.	n-Propane	secondary	2.11e-09	1.49e-06
Japanese Electric Grid	Benzo[a]anthracene	model/secondary	2.05e-09	1.45e-06
US electric grid	Acetophenone	model/secondary	1.98e-09	1.40e-06
LPG Production	Dichloromethane	secondary	1.96e-09	1.38e-06
US electric grid	Methyl chloride	model/secondary	1.93e-09	1.37e-06
LPG Production	Aromatic hydrocarbons	secondary	1.92e-09	1.36e-06
Japanese Electric Grid	Ethylene dibromide	model/secondary	1.84e-09	1.30e-06
Japanese Electric Grid	Chlorobenzene	model/secondary	1.78e-09	1.26e-06
Fuel Oil #4 Prod.	Isophorone	secondary	1.52e-09	1.08e-06
Japanese Electric Grid	Chrysene	model/secondary	1.37e-09	9.68e-07
Natural Gas Prod	Ethylbenzene	secondary	1 32e-09	9 37e-07
LPG Production	Cumene	secondary	1.29e-09	9.15e-07
Japanese Electric Grid	Anthracene	model/secondary	1.17e-09	8.24e-07
Natural Gas Prod.	Bromomethane	secondary	1.17e-09	8.24e-07
Japanese Electric Grid	Indeno(1.2.3-cd)pyrene	model/secondary	1.15e-09	8.10e-07
Japanese Electric Grid	Benzolg.h.ilpervlene	model/secondary	1.10e-09	7.78e-07
US electric grid	Dichloromethane	model/secondary	1.06e-09	7.48e-07
Japanese Electric Grid	Benzo[b.i.k]fluoranthene	model/secondary	9.94e-10	7.03e-07
Fuel Oil #4 Prod.	Toluene	secondary	9.81e-10	6.94e-07
Fuel Oil #4 Prod	Acetaldehyde	secondary	8 71e-10	6 16e-07
Japanese Electric Grid	A cenanbthylene	model/secondary	8 43e-10	5.96e-07
LPG Production	Phenanthrene	secondary	8 28e-10	5.85e-07
Japanese Electric Grid	2 4-Dinitrotoluene	model/secondary	8.17e-10	5.05 c 07
Fuel Oil #6 Prod	Isonhorone	secondary	8.11e-10	5.74e-07
Fuel Oil #4 Prod	Propionaldehyde	secondary	7 90e-10	5.59e-07
US electric grid	Cumene	model/secondary	7.00e-10	4.95e-07
Fuel Oil #4 Prod	Methyl ethyl ketone	secondary	6 36e-10	4.50e-07
Japanese Electric Grid	2-Methylnanhthalene	model/secondary	6.09e-10	4.30e-07
Japanese Electric Grid	2 Chloroscetonhenone	model/secondary	5.65e 10	4.00e.07
LPG Production	Vinyl acetate	secondary	5.03C-10	3.850.07
Ero Floduction	Toluene	secondary	J.440-10	3 300 07
Fuel Oil #6 Prod	A cetaldehyde	secondary	4.600-10	3 280 07
Fuel Oil #4 Prod	A crolein	secondary	4.040-10	3.120.07
Fuel Oil #2 Prod	Isonhorona	secondary	4.430-10	3.060.07
Fuel Oil #6 Brod	Dronionaldahuda	secondary	4.326-10	2.00-07
LPG Production	Diphonyl	secondary	4.21e-10	2.966-07
LIG Froduction	Chloroform	secondary	4.15e-10	2.946-07
	Chiorofolini	secondary	5.900-10	2.010-07

 Table M-18.
 LCD LCIA Results for the Photochemical Smog Impact Category

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
US electric grid	Phenanthrene	model/secondary	3.69e-10	2.61e-07
Fuel Oil #4 Prod.	n-Propane	secondary	3.60e-10	2.55e-07
Fuel Oil #4 Prod.	o-xylene	secondary	3.55e-10	2.51e-07
Natural Gas Prod.	Styrene	secondary	3.48e-10	2.46e-07
Fuel Oil #6 Prod.	Methyl ethyl ketone	secondary	3.39e-10	2.40e-07
LPG Production	1,4-Dichlorobenzene	secondary	3.03e-10	2.14e-07
Fuel Oil #2 Prod.	Toluene	secondary	2.97e-10	2.10e-07
US electric grid	Vinyl acetate	model/secondary	2.94e-10	2.08e-07
LPG Production	Tetrachloroethylene	secondary	2.90e-10	2.05e-07
LPG Production	Ethyl Chloride	secondary	2.83e-10	2.00e-07
LPG Production	1,2-Dichloroethane	secondary	2.70e-10	1.91e-07
Natural Gas Prod.	Benzyl chloride	secondary	2.69e-10	1.90e-07
Natural Gas Prod.	Methyl tert-butyl ether	secondary	2.55e-10	1.80e-07
LPG Production	Fluorene	secondary	2.53e-10	1.79e-07
Fuel Oil #4 Prod.	Ethylbenzene	secondary	2.49e-10	1.76e-07
Fuel Oil #2 Prod.	Acetaldehvde	secondary	2.47e-10	1.75e-07
Fuel Oil #6 Prod.	Acrolein	secondary	2.36e-10	1.67e-07
US electric grid	Biphenyl	model/secondary	2.25e-10	1.59e-07
Fuel Oil #2 Prod.	Propionaldehyde	secondary	2.24e-10	1.59e-07
Natural Gas Prod	Phenol	secondary	2.23e-10	1 58e-07
Fuel Oil #4 Prod	Bromomethane	secondary	2.20e-10	1.55e-07
LPG Production	2-Methylnaphthalene	secondary	2.19e-10	1.55e-07
US electric grid	Chloroform	model/secondary	2.15e-10	1.550 07
Natural Gas Prod	Acetophenone	secondary	2.09e-10	1.32e 07
LPG Production	Fluoranthene	secondary	2.07e-10	1.16e-07
Natural Gas Prod	Methyl chloride	secondary	2.07e-10	1.44e-07
LPG Production	Acenaphthene	secondary	1.92e-10	1.35e-07
Fuel Oil #6 Prod	n-Propane	secondary	1.92e-10	1.35e-07
Fuel Oil #2 Prod	Methyl ethyl ketone	secondary	1.91e-10	1.35e 07
Fuel Oil #6 Prod	o-xylene	secondary	1.67e-10	1.20c 07
US electric grid	Tetrachloroethylene	model/secondary	1.57e-10	1.10C-07
I PG Production	Ethylene dibromide	secondary	1.576-10	1.08e.07
US electric grid	Ethyl Chloride	model/secondary	1.53e-10	1.080-07
L DG Production	Chlorabanzana	socondary	1.330-10	1.050.07
LFO Floadetion	1.2 Dicklareathana	medal/accordamy	1.460-10	1.03e-07
L BG Broduction	1,2-Dichloroethane	model/secondary	1.400-10	0.520.08
Erel Oil #6 Prod	Ethylhongong	secondary	1.336-10	9.556-08
L DC Drackastica	Euryidenzene	secondary	1.326-10	9.336-08
Erel Oil #2 Prod	A surplain	secondary	1.31e-10	9.29e-08
Fuel Oil #2 Prod.	Acrolein	secondary	1.26e-10	8.90e-08
US electric grid	Fluorene	model/secondary	1.24e-10	8.75e-08
Fuel Oil #4 Prod.	Naphthalene	secondary	1.18e-10	8.36e-08
Fuel Oil #6 Prod.	Bromomethane	secondary	1.1/e-10	8.28e-08
Natural Gas Prod.	Dichloromethane	secondary	1.11e-10	7.88e-08
Japanese Electric Grid	Benzo[a]pyrene	model/secondary	1.11e-10	7.87e-08
Fuel Oil #2 Prod.	o-xylene	secondary	1.10e-10	7.80e-08
Fuel Oil #2 Prod.	n-Propane	secondary	1.03e-10	7.25e-08
US electric grid	Fluoranthene	model/secondary	9.89e-11	6.99e-08
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	8.95e-11	6.33e-08
Natural Gas Prod.	Phenanthrene	secondary	8.45e-11	5.98e-08
US electric grid	o-xylene	model/secondary	8.41e-11	5.94e-08

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
US electric grid	Acenaphthene	model/secondary	8.37e-11	5.92e-08
US electric grid	Ethylene dibromide	model/secondary	8.29e-11	5.86e-08
US electric grid	Chlorobenzene	model/secondary	8.02e-11	5.67e-08
LPG Production	Acenaphthylene	secondary	7.81e-11	5.52e-08
US electric grid	1,1,1-Trichloroethane	model/secondary	7.80e-11	5.51e-08
LPG Production	Anthracene	secondary	7.62e-11	5.38e-08
Natural Gas Prod.	Cumene	secondary	7.38e-11	5.22e-08
Fuel Oil #2 Prod.	Ethylbenzene	secondary	7.07e-11	5.00e-08
LPG Production	2,4-Dinitrotoluene	secondary	6.84e-11	4.84e-08
Fuel Oil #4 Prod.	Styrene	secondary	6.56e-11	4.64e-08
Natural Gas Prod.	2-Methylnaphthalene	secondary	6.49e-11	4.59e-08
Japanese Electric Grid	5-Methyl chrysene	model/secondary	6.43e-11	4.55e-08
Fuel Oil #2 Prod.	Bromomethane	secondary	6.24e-11	4.41e-08
Fuel Oil #6 Prod.	Naphthalene	secondary	5.48e-11	3.87e-08
Fuel Oil #4 Prod.	Benzvl chloride	secondary	5.07e-11	3.58e-08
US electric grid	Pvrene	model/secondary	4.91e-11	3.47e-08
Fuel Oil #4 Prod	Methyl tert-butyl ether	secondary	4 80e-11	3 40e-08
LPG Production	2-Chloroacetophenone	secondary	4 72e-11	3 34e-08
LPG Production	Benzo[a]anthracene	secondary	4 56e-11	3 22e-08
LPG Production	Chrysene	secondary	4.30e-11	3.09e-08
Fuel Oil #4 Prod	Phenol	secondary	4.37e 11	2.97e-08
Fuel Oil #4 Prod	Acetophenone	secondary	3.94e-11	2.57e-08
Fuel Oil #4 Prod	Methyl chloride	secondary	3.84e-11	2.78e-08
Fuel Oil #2 Prod	Nanhthalene	secondary	3.71e-11	2.710-08
LIS algorita grid	2.4 Dinitrotoluono	model/cacendary	2 702 11	2.620-08
Fuel Oil #6 Prod	2,4-Dimuotoluciic	secondary	2 500 11	2.020-08
L PG Production	Indono(1.2.2. ad)nyrona	secondary	2 200 11	2.476-08
LEO Floateia grid	A conceptbylone	model/accordany	2 220 11	2.406-08
Natural Gas Prod	Vinyl acetate	secondary	3.320-11	2.356-08
Natural Gas Flou.	A mthreasons	medal/accordamy	2.870.11	2.196-08
Evel Oil #C Prod	Anurracene Devend shlewide	model/secondary	2.876-11	2.03e-08
Fuel OII #6 Prod.	Benzyl chloride	secondary	2.70e-11	1.91e-08
LPG Production	Benzo[b,j,k]fluorantnene	secondary	2.69e-11	1.90e-08
	Methyl tert-butyl ether	secondary	2.50e-11	1.81e-08
US electric grid	2-Chloroacetophenone	model/secondary	2.55e-11	1.80e-08
LPG Production	Benzo[a]pyrene	secondary	2.50e-11	1.76e-08
Natural Gas Prod.	Bipnenyi	secondary	2.37e-11	1.67e-08
Natural Gas Prod.	Chloroform	secondary	2.2/e-11	1.60e-08
Fuel Oil #6 Prod.	Phenol	secondary	2.24e-11	1.58e-08
Fuel Oil #4 Prod.	Dichloromethane	secondary	2.10e-11	1.48e-08
Fuel Oil #6 Prod.	Acetophenone	secondary	2.10e-11	1.48e-08
Natural Gas Prod.	Fluorene	secondary	2.07e-11	1.46e-08
Fuel Oil #6 Prod.	Methyl chloride	secondary	2.05e-11	1.45e-08
LPG Production	Benzo[g,h,i]perylene	secondary	2.03e-11	1.43e-08
Fuel Oil #2 Prod.	Styrene	secondary	1.86e-11	1.32e-08
Natural Gas Prod.	Pyrene	secondary	1.85e-11	1.31e-08
Natural Gas Prod.	Fluoranthene	secondary	1.84e-11	1.30e-08
Natural Gas Prod.	Tetrachloroethylene	secondary	1.65e-11	1.17e-08
Natural Gas Prod.	Ethyl Chloride	secondary	1.61e-11	1.14e-08

 Table M-18. LCD LCIA Results for the Photochemical Smog Impact Category

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Natural Gas Prod.	Acenaphthene	secondary	1.59e-11	1.12e-08
US electric grid	Benzo[b,j,k]fluoranthene	model/secondary	1.57e-11	1.11e-08
Natural Gas Prod.	1,2-Dichloroethane	secondary	1.54e-11	1.09e-08
US electric grid	Chrysene	model/secondary	1.51e-11	1.06e-08
Fuel Oil #2 Prod.	Benzyl chloride	secondary	1.44e-11	1.02e-08
Fuel Oil #4 Prod.	Aromatic hydrocarbons	secondary	1.40e-11	9.93e-09
Fuel Oil #4 Prod.	Cumene	secondary	1.39e-11	9.83e-09
US electric grid	Benzo[a]anthracene	model/secondary	1.37e-11	9.66e-09
Fuel Oil #2 Prod.	Methyl tert-butyl ether	secondary	1.36e-11	9.65e-09
Fuel Oil #2 Prod.	Phenol	secondary	1.19e-11	8.43e-09
Fuel Oil #6 Prod.	Dichloromethane	secondary	1.12e-11	7.91e-09
Fuel Oil #2 Prod.	Acetophenone	secondary	1.12e-11	7.91e-09
Fuel Oil #2 Prod.	Methyl chloride	secondary	1.09e-11	7.71e-09
US electric grid	Indeno(1,2,3-cd)pyrene	model/secondary	9.71e-12	6.87e-09
Natural Gas Prod.	Anthracene	secondary	9.64e-12	6.81e-09
Natural Gas Prod.	Ethylene dibromide	secondary	8.74e-12	6.18e-09
Fuel Oil #4 Prod.	Phenanthrene	secondary	8.58e-12	6.07e-09
Natural Gas Prod.	Chlorobenzene	secondary	8.45e-12	5.98e-09
Natural Gas Prod.	Acenaphthylene	secondary	8.39e-12	5.93e-09
Natural Gas Prod.	1.1.1-Trichloroethane	secondary	7.68e-12	5.43e-09
Fuel Oil #6 Prod.	Cumene	secondary	7.41e-12	5.24e-09
Natural Gas Prod.	Benzolalanthracene	secondary	6.72e-12	4.75e-09
Natural Gas Prod.	Chrysene	secondary	6.48e-12	4.58e-09
Fuel Oil #2 Prod.	Dichloromethane	secondary	5.97e-12	4.22e-09
Natural Gas Prod.	Indeno(1.2.3-cd)pyrene	secondary	5.91e-12	4.18e-09
Fuel Oil #4 Prod.	Vinvl acetate	secondary	5.85e-12	4.13e-09
LPG Production	5-Methyl chrysene	secondary	5.38e-12	3.80e-09
US electric grid	Benzolg h ilnervlene	model/secondary	5 31e-12	3 75e-09
Fuel Oil #2 Prod.	Aromatic hydrocarbons	secondary	5.23e-12	3.70e-09
US electric grid	Benzo[a]pyrene	model/secondary	5.02e-12	3.55e-09
Fuel Oil #6 Prod.	Aromatic hydrocarbons	secondary	4.60e-12	3.25e-09
Fuel Oil #4 Prod	Binhenvl	secondary	4 46e-12	3 15e-09
Fuel Oil #6 Prod	Phenanthrene	secondary	4 44e-12	3 14e-09
Fuel Oil #4 Prod	Chloroform	secondary	4 27e-12	3 02e-09
Fuel Oil #2 Prod	Cumene	secondary	3.95e-12	2 79e-09
US electric orid	2-Methylnanhthalene	model/secondary	3.93e-12	2.79e-09
Natural Gas Prod	2 4-Dinitrotoluene	secondary	3.90e-12	2.76e-09
Natural Gas Prod	Benzola h ilpervlene	secondary	3.83e-12	2.70e 09
Natural Gas Prod	Benzo[a]nvrene	secondary	3 78e-12	2.71e-09
Fuel Oil #6 Prod	Vinvl acetate	secondary	3 12e-12	2.07e-09
Fuel Oil #4 Prod	Tetrachloroethylene	secondary	3 11e 12	2.200-09
Fuel Oil #4 Prod	Ethyl Chloride	secondary	3.04e 12	2.200-09
US electric grid	5 Mathul abrusana	model/secondamy	2.010.12	2.150-09
Fuel Oil #4 Prod	1.2 Diablereathana	secondary	2.916-12	2.000-09
Fuel Oil #4 Prod.	1,2-Dichlorobenzene	secondary	2.906-12	1.030.00
Natural Gas Prod	2 Chlorogestonhanene	secondary	2.750-12	1.950-09
Eval Oil #4 Drod	2-Chioroacetophenone	secondary	2.096-12	1.900-09
Fuel Oil #4 Prod.	F IUOFEIIC	secondary	2.000-12	1.886-09
Fuel Oil #2 Prod	r nenanulirene	secondary	2.50e-12	1./00-09
	CI C	secondary	2.38e-12	1.686-09
Fuel Oil #6 Prod.	Chlorotorm	secondary	2.28e-12	1.61e-09

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Fuel Oil #4 Prod.	Fluoranthene	secondary	2.15e-12	1.52e-09
LPG Production	Polycyclic aromatic hydrocarbons	secondary	2.13e-12	1.51e-09
Fuel Oil #4 Prod.	2-Methylnaphthalene	secondary	1.98e-12	1.40e-09
Fuel Oil #4 Prod.	Acenaphthene	secondary	1.86e-12	1.31e-09
Fuel Oil #2 Prod.	Vinyl acetate	secondary	1.66e-12	1.17e-09
Fuel Oil #6 Prod.	Tetrachloroethylene	secondary	1.66e-12	1.17e-09
Fuel Oil #4 Prod.	Ethylene dibromide	secondary	1.65e-12	1.16e-09
Fuel Oil #6 Prod.	Ethyl Chloride	secondary	1.62e-12	1.15e-09
Fuel Oil #4 Prod.	Chlorobenzene	secondary	1.59e-12	1.13e-09
Fuel Oil #6 Prod.	1,2-Dichloroethane	secondary	1.54e-12	1.09e-09
Natural Gas Prod.	Benzo[b,j,k]fluoranthene	secondary	1.53e-12	1.08e-09
Fuel Oil #4 Prod.	1,1,1-Trichloroethane	secondary	1.45e-12	1.02e-09
Fuel Oil #6 Prod.	Fluorene	secondary	1.39e-12	9.83e-10
Fuel Oil #4 Prod.	Pyrene	secondary	1.32e-12	9.30e-10
Natural Gas Prod.	Aromatic hydrocarbons	secondary	1.27e-12	9.00e-10
Fuel Oil #2 Prod.	Biphenyl	secondary	1.27e-12	8.96e-10
Fuel Oil #6 Prod.	1,4-Dichlorobenzene	secondary	1.23e-12	8.68e-10
Fuel Oil #2 Prod.	Chloroform	secondary	1.21e-12	8.58e-10
Fuel Oil #6 Prod.	Fluoranthene	secondary	1.12e-12	7.91e-10
Fuel Oil #6 Prod.	Acenaphthene	secondary	9.03e-13	6.39e-10
Fuel Oil #6 Prod.	2-Methylnaphthalene	secondary	8.89e-13	6.29e-10
Fuel Oil #2 Prod.	Tetrachloroethylene	secondary	8.85e-13	6.25e-10
Fuel Oil #6 Prod.	Ethylene dibromide	secondary	8.78e-13	6.21e-10
Fuel Oil #2 Prod.	1,4-Dichlorobenzene	secondary	8.74e-13	6.18e-10
Fuel Oil #2 Prod.	Ethyl Chloride	secondary	8.64e-13	6.11e-10
Fuel Oil #6 Prod.	Chlorobenzene	secondary	8.49e-13	6.00e-10
Fuel Oil #2 Prod.	1,2-Dichloroethane	secondary	8.23e-13	5.82e-10
Fuel Oil #4 Prod.	Acenaphthylene	secondary	8.09e-13	5.72e-10
Fuel Oil #6 Prod.	1,1,1-Trichloroethane	secondary	7.72e-13	5.46e-10
Fuel Oil #4 Prod.	Anthracene	secondary	7.71e-13	5.45e-10
Fuel Oil #2 Prod.	Fluorene	secondary	7.67e-13	5.42e-10
Fuel Oil #4 Prod.	2.4-Dinitrotoluene	secondary	7.35e-13	5.20e-10
Fuel Oil #6 Prod.	Pyrene	secondary	6.59e-13	4.66e-10
Fuel Oil #2 Prod.	2-Methylnaphthalene	secondary	6.34e-13	4.48e-10
Fuel Oil #2 Prod.	Fluoranthene	secondary	6.24e-13	4.41e-10
Fuel Oil #2 Prod.	Acenaphthene	secondary	5.66e-13	4.00e-10
Fuel Oil #4 Prod	2-Chloroacetophenone	secondary	5.07e-13	3 58e-10
Fuel Oil #2 Prod	Ethylene dibromide	secondary	4 68e-13	3 31e-10
Fuel Oil #2 Prod	Chlorobenzene	secondary	4 53e-13	3 20e-10
Fuel Oil #4 Prod	Chrysene	secondary	4 32e-13	3 05e-10
Fuel Oil #4 Prod	Benzo[a]anthracene	secondary	4 29e-13	3.03e-10
Fuel Oil #6 Prod	A cenaphthylene	secondary	4 18e-13	2.95e-10
Fuel Oil #2 Prod	1 1 1-Trichloroethane	secondary	4 11e-13	2.91e-10
Fuel Oil #2 Prod	Pyrene	secondary	3.92e-13	2.77e-10
Fuel Oil #6 Prod	2 4-Dinitrotoluene	secondary	3.92e-13	2.77e-10
Fuel Oil #6 Prod	Anthracene	secondary	3.90e-13	2.776-10
Fuel Oil #4 Prod	Indeno(1.2.3-cd)pyrene	secondary	3.27e-13	2.700 10 2.31e-10
Natural Gas Prod	5-Methyl chrysene	secondary	3.06e-13	2.17e-10
		occontail y	5.000 15	2.170 10

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Fuel Oil #4 Prod.	Benzo[b,j,k]fluoranthene	secondary	2.89e-13	2.04e-10
Fuel Oil #6 Prod.	2-Chloroacetophenone	secondary	2.70e-13	1.91e-10
Fuel Oil #4 Prod.	Benzo[a]pyrene	secondary	2.57e-13	1.82e-10
Fuel Oil #2 Prod.	Acenaphthylene	secondary	2.36e-13	1.67e-10
Fuel Oil #2 Prod.	Anthracene	secondary	2.28e-13	1.61e-10
Fuel Oil #6 Prod.	Chrysene	secondary	2.14e-13	1.51e-10
Fuel Oil #2 Prod.	2,4-Dinitrotoluene	secondary	2.09e-13	1.48e-10
Fuel Oil #6 Prod.	Benzo[a]anthracene	secondary	2.02e-13	1.43e-10
Fuel Oil #4 Prod.	Benzo[g,h,i]perylene	secondary	1.90e-13	1.34e-10
Fuel Oil #6 Prod.	Indeno(1,2,3-cd)pyrene	secondary	1.58e-13	1.12e-10
Fuel Oil #6 Prod.	Benzo[b,j,k]fluoranthene	secondary	1.54e-13	1.09e-10
Fuel Oil #2 Prod.	2-Chloroacetophenone	secondary	1.44e-13	1.02e-10
Fuel Oil #2 Prod.	Benzo[a]anthracene	secondary	1.33e-13	9.43e-11
Fuel Oil #6 Prod.	Benzo[a]pyrene	secondary	1.32e-13	9.35e-11
Fuel Oil #2 Prod.	Chrysene	secondary	1.30e-13	9.18e-11
Fuel Oil #2 Prod.	Indeno(1,2,3-cd)pyrene	secondary	1.00e-13	7.07e-11
Fuel Oil #6 Prod.	Benzo[g,h,i]perylene	secondary	8.88e-14	6.28e-11
Fuel Oil #2 Prod.	Benzo[b,j,k]fluoranthene	secondary	8.20e-14	5.80e-11
Fuel Oil #2 Prod.	Benzo[a]pyrene	secondary	7.51e-14	5.31e-11
Fuel Oil #2 Prod.	Benzo[g,h,i]perylene	secondary	5.92e-14	4.19e-11
Fuel Oil #4 Prod.	5-Methyl chrysene	secondary	5.77e-14	4.08e-11
Fuel Oil #6 Prod.	5-Methyl chrysene	secondary	3.08e-14	2.18e-11
Fuel Oil #2 Prod.	5-Methyl chrysene	secondary	1.64e-14	1.16e-11
Fuel Oil #4 Prod.	Polycyclic aromatic hydrocarbons	secondary	1.56e-14	1.10e-11
Fuel Oil #2 Prod.	Polycyclic aromatic hydrocarbons	secondary	5.82e-15	4.11e-12
Fuel Oil #6 Prod.	Polycyclic aromatic hydrocarbons	secondary	5.11e-15	3.62e-12
Natural Gas Prod.	Polycyclic aromatic hydrocarbons	secondary	1.42e-15	1.00e-12
Total Manufacturing			1.18e-02	8.33e+00
Use, Maintenance and Repair L	life-cycle Stage			
US electric grid	Methane	model/secondary	1.68e-03	1.19e+00
US electric grid	Isophorone	model/secondary	1.48e-05	1.04e-02
US electric grid	Formaldehyde	model/secondary	1.33e-05	9.40e-03
US electric grid	Acetaldehyde	model/secondary	8.45e-06	5.98e-03
US electric grid	Benzene	model/secondary	8.23e-06	5.82e-03
US electric grid	Propionaldehyde	model/secondary	7.67e-06	5.42e-03
US electric grid	Methyl ethyl ketone	model/secondary	6.17e-06	4.37e-03
US electric grid	Toluene	model/secondary	5.34e-06	3.78e-03
US electric grid	Acrolein	model/secondary	4.30e-06	3.04e-03
US electric grid	Ethylbenzene	model/secondary	2.40e-06	1.70e-03
US electric grid	Bromomethane	model/secondary	2.13e-06	1.51e-03
US electric grid	Xylene (mixed isomers)	model/secondary	9.42e-07	6.66e-04
US electric grid	Hexane	model/secondary	9.33e-07	6.60e-04
US electric grid	Styrene	model/secondary	6.37e-07	4.50e-04
US electric grid	Naphthalene	model/secondary	5.19e-07	3.67e-04
US electric grid	Benzyl chloride	model/secondary	4.92e-07	3.48e-04
US electric grid	Methyl tert-butyl ether	model/secondary	4.66e-07	3.30e-04
US electric grid	Phenol	model/secondary	4.08e-07	2.88e-04
US electric grid	Acetophenone	model/secondary	3.82e-07	2.70e-04
US electric grid	Methyl chloride	model/secondary	3.73e-07	2.63e-04

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
US electric grid	Dichloromethane	model/secondary	2.04e-07	1.44e-04
US electric grid	Cumene	model/secondary	1.35e-07	9.54e-05
US electric grid	Phenanthrene	model/secondary	7.12e-08	5.03e-05
US electric grid	Vinyl acetate	model/secondary	5.67e-08	4.01e-05
US electric grid	Biphenyl	model/secondary	4.33e-08	3.06e-05
US electric grid	Chloroform	model/secondary	4.15e-08	2.93e-05
US electric grid	Tetrachloroethylene	model/secondary	3.02e-08	2.14e-05
US electric grid	Ethyl Chloride	model/secondary	2.95e-08	2.09e-05
US electric grid	1,2-Dichloroethane	model/secondary	2.81e-08	1.99e-05
US electric grid	Fluorene	model/secondary	2.38e-08	1.69e-05
US electric grid	Fluoranthene	model/secondary	1.91e-08	1.35e-05
US electric grid	o-xylene	model/secondary	1.62e-08	1.15e-05
US electric grid	Acenaphthene	model/secondary	1.61e-08	1.14e-05
US electric grid	Ethylene dibromide	model/secondary	1.60e-08	1.13e-05
US electric grid	Chlorobenzene	model/secondary	1.55e-08	1.09e-05
US electric grid	1,1,1-Trichloroethane	model/secondary	1.50e-08	1.06e-05
US electric grid	Pyrene	model/secondary	9.46e-09	6.69e-06
US electric grid	2,4-Dinitrotoluene	model/secondary	7.13e-09	5.04e-06
US electric grid	Acenaphthylene	model/secondary	6.41e-09	4.53e-06
US electric grid	Anthracene	model/secondary	5.53e-09	3.91e-06
US electric grid	2-Chloroacetophenone	model/secondary	4.92e-09	3.48e-06
US electric grid	Benzo[b,j,k]fluoranthene	model/secondary	3.02e-09	2.14e-06
US electric grid	Chrysene	model/secondary	2.90e-09	2.05e-06
US electric grid	Benzo[a]anthracene	model/secondary	2.63e-09	1.86e-06
US electric grid	Indeno(1,2,3-cd)pyrene	model/secondary	1.87e-09	1.32e-06
US electric grid	Benzolg.h.ilpervlene	model/secondary	1.02e-09	7.24e-07
US electric grid	Benzo[a]pyrene	model/secondary	9.68e-10	6.84e-07
US electric grid	2-Methylnaphthalene	model/secondary	7.57e-10	5.36e-07
US electric grid	5-Methyl chrysene	model/secondary	5.60e-10	3.96e-07
Total Use. Maintenance and Re	nair	,	1 760-03	1 250+00
End of life Life and Stand	Pan		1.700-03	1.230+00
Lop 1 191	TT 1 1 1 1 1 1 1 1 1 1	•	2.59 .05	1.02.02
	Hydrocarbons, remaining unspeciated	primary	2.58e-05	1.83e-02
US electric grid	Methane	model/secondary	3.20e-07	2.26e-04
LPG Production	Hydrocarbons, remaining unspeciated	secondary	2.45e-07	1.73e-04
LPG Production	Nonmethane hydrocarbons, remaining unspeciated	secondary	1.6/e-0/	1.18e-04
LCD landfilling	Methane	primary	9.18e-08	6.49e-05
LCD landfilling	Benzene	primary	3.08e-08	2.18e-05
LPG Production	Methane	secondary	2.31e-08	1.64e-05
LPG Production	Benzene	secondary	1.07e-08	7.55e-06
LCD landfilling	Toluene	primary	6.48e-09	4.58e-06
LCD landfilling	Xylene (mixed isomers)	primary	4.54e-09	3.21e-06
US electric grid	Isophorone	model/secondary	2.80e-09	1.98e-06
LPG Production	Aldehydes	secondary	2.61e-09	1.84e-06
US electric grid	Formaldehyde	model/secondary	2.52e-09	1.78e-06
LPG Production	Formaldehyde	secondary	1.94e-09	1.37e-06
US electric grid	Acetaldehyde	model/secondary	1.60e-09	1.13e-06
US electric grid	Benzene	model/secondary	1.56e-09	1.10e-06
US electric grid	Propionaldehyde	model/secondary	1.46e-09	1.03e-06
LPG Production	Ethane	secondary	1.22e-09	8.60e-07

 Table M-18.
 LCD LCIA Results for the Photochemical Smog Impact Category

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
US electric grid	Methyl ethyl ketone	model/secondary	1.17e-09	8.28e-07
LPG Production	Pentane	secondary	1.05e-09	7.39e-07
US electric grid	Toluene	model/secondary	1.01e-09	7.17e-07
LCD landfilling	Ethylbenzene	primary	9.02e-10	6.38e-07
LPG Production	Butane	secondary	8.24e-10	5.82e-07
US electric grid	Acrolein	model/secondary	8.16e-10	5.77e-07
LPG Production	Hexane	secondary	7.39e-10	5.22e-07
US electric grid	Ethylbenzene	model/secondary	4.56e-10	3.22e-07
US electric grid	Bromomethane	model/secondary	4.04e-10	2.86e-07
US electric grid	Xylene (mixed isomers)	model/secondary	1.79e-10	1.26e-07
US electric grid	Hexane	model/secondary	1.77e-10	1.25e-07
US electric grid	Styrene	model/secondary	1.21e-10	8.54e-08
US electric grid	Naphthalene	model/secondary	9.85e-11	6.97e-08
US electric grid	Benzyl chloride	model/secondary	9.34e-11	6.60e-08
US electric grid	Methyl tert-butyl ether	model/secondary	8.85e-11	6.25e-08
US electric grid	Phenol	model/secondary	7.73e-11	5.47e-08
US electric grid	Acetophenone	model/secondary	7.25e-11	5.13e-08
US electric grid	Methyl chloride	model/secondary	7.07e-11	5.00e-08
US electric grid	Dichloromethane	model/secondary	3.87e-11	2.73e-08
US electric grid	Cumene	model/secondary	2.56e-11	1.81e-08
LCD landfilling	Trichloroethylene	primary	2.18e-11	1.54e-08
LCD incineration	Trichloroethylene	secondary	1.69e-11	1.19e-08
LCD landfilling	Vinyl chloride	primary	1.38e-11	9.78e-09
US electric grid	Phenanthrene	model/secondary	1.35e-11	9.55e-09
LPG Production	Isophorone	secondary	1.16e-11	8.22e-09
LCD landfilling	Dichloromethane	primary	1.11e-11	7.82e-09
US electric grid	Vinyl acetate	model/secondary	1.08e-11	7.61e-09
LCD incineration	Vinyl chloride	secondary	1.07e-11	7.59e-09
LPG Production	Toluene	secondary	8.24e-12	5.83e-09
US electric grid	Biphenyl	model/secondary	8.22e-12	5.81e-09
US electric grid	Chloroform	model/secondary	7.87e-12	5.56e-09
LCD landfilling	1,2-Dichloroethane	primary	6.92e-12	4.90e-09
LCD landfilling	Carbon tetrachloride	primary	6.92e-12	4.90e-09
LCD landfilling	Chloroform	primary	6.92e-12	4.90e-09
LCD landfilling	Tetrachloroethylene	primary	6.92e-12	4.90e-09
LPG Production	Acetaldehyde	secondary	6.66e-12	4.71e-09
LPG Production	Propionaldehyde	secondary	6.04e-12	4.27e-09
US electric grid	Tetrachloroethylene	model/secondary	5.73e-12	4.05e-09
US electric grid	Ethyl Chloride	model/secondary	5.60e-12	3.96e-09
LCD incineration	Carbon tetrachloride	secondary	5.37e-12	3.80e-09
US electric grid	1,2-Dichloroethane	model/secondary	5.33e-12	3.77e-09
LPG Production	Methyl ethyl ketone	secondary	4.86e-12	3.44e-09
US electric grid	Fluorene	model/secondary	4.52e-12	3.20e-09
US electric grid	Fluoranthene	model/secondary	3.62e-12	2.56e-09
LPG Production	Acrolein	secondary	3.39e-12	2.39e-09
LPG Production	o-xylene	secondary	3.09e-12	2.19e-09
US electric grid	o-xylene	model/secondary	3.08e-12	2.17e-09
US electric grid	Acenaphthene	model/secondary	3.06e-12	2.16e-09
US electric grid	Ethylene dibromide	model/secondary	3.03e-12	2.14e-09
US electric grid	Chlorobenzene	model/secondary	2.93e-12	2.07e-09

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
US electric grid	1,1,1-Trichloroethane	model/secondary	2.85e-12	2.02e-09
LPG Production	n-Propane	secondary	2.76e-12	1.95e-09
LPG Production	Ethylbenzene	secondary	1.90e-12	1.35e-09
US electric grid	Pyrene	model/secondary	1.79e-12	1.27e-09
LPG Production	Bromomethane	secondary	1.68e-12	1.19e-09
US electric grid	2,4-Dinitrotoluene	model/secondary	1.35e-12	9.57e-10
US electric grid	Acenaphthylene	model/secondary	1.22e-12	8.59e-10
US electric grid	Anthracene	model/secondary	1.05e-12	7.42e-10
LPG Production	Naphthalene	secondary	1.05e-12	7.40e-10
US electric grid	2-Chloroacetophenone	model/secondary	9.34e-13	6.60e-10
US electric grid	Benzo[b,i,k]fluoranthene	model/secondary	5.73e-13	4.05e-10
US electric grid	Chrysene	model/secondary	5 50e-13	3 89e-10
LPG Production	Styrene	secondary	5.01e-13	3 54e-10
US electric grid	Benzo[a]anthracene	model/secondary	5 00e-13	3 53e-10
LPG Production	Benzyl chloride	secondary	3.87e-13	2 74e-10
LPG Production	Methyl tert-butyl ether	secondary	3.67e-13	2.74e 10
LIS electric grid	Indeno(1,2,3, cd)pyrana	model/secondary	3.55e 13	2.00e-10
LPG Production	Phonol	secondary	2 210 12	2.316-10
LPC Production	Agetenhenene	secondary	2.012.12	2.276-10
LPG Production	Acetophenone	secondary	3.01e-13	2.13e-10
LPG Production	Methyl chloride	secondary	2.93e-13	2.07e-10
US electric grid	Benzo[g,n,1]perylene	model/secondary	1.94e-13	1.3/e-10
US electric grid	Benzo[a]pyrene	model/secondary	1.84e-13	1.30e-10
LPG Production	Dichloromethane	secondary	1.61e-13	1.13e-10
LPG Production	Aromatic hydrocarbons	secondary	1.57e-13	1.11e-10
US electric grid	2-Methylnaphthalene	model/secondary	1.44e-13	1.02e-10
US electric grid	5-Methyl chrysene	model/secondary	1.06e-13	7.52e-11
LPG Production	Cumene	secondary	1.06e-13	7.52e-11
LPG Production	Phenanthrene	secondary	6.79e-14	4.80e-11
LPG Production	Vinyl acetate	secondary	4.47e-14	3.16e-11
LPG Production	Biphenyl	secondary	3.41e-14	2.41e-11
LPG Production	Chloroform	secondary	3.27e-14	2.31e-11
LPG Production	1,4-Dichlorobenzene	secondary	2.48e-14	1.76e-11
LPG Production	Tetrachloroethylene	secondary	2.38e-14	1.68e-11
LPG Production	Ethyl Chloride	secondary	2.32e-14	1.64e-11
LPG Production	1,2-Dichloroethane	secondary	2.21e-14	1.57e-11
LPG Production	Fluorene	secondary	2.08e-14	1.47e-11
LPG Production	2-Methylnaphthalene	secondary	1.80e-14	1.27e-11
LPG Production	Fluoranthene	secondary	1.70e-14	1.20e-11
LPG Production	Acenaphthene	secondary	1.57e-14	1.11e-11
LPG Production	Ethylene dibromide	secondary	1.26e-14	8.90e-12
LPG Production	Chlorobenzene	secondary	1.22e-14	8.61e-12
LPG Production	1,1,1-Trichloroethane	secondary	1.11e-14	7.83e-12
LPG Production	Pyrene	secondary	1.08e-14	7.63e-12
LPG Production	Acenaphthylene	secondary	6.41e-15	4.53e-12
LPG Production	Anthracene	secondary	6.25e-15	4.42e-12
LPG Production	2,4-Dinitrotoluene	secondary	5.62e-15	3.97e-12
LPG Production	2-Chloroacetonhenone	secondary	3 87e-15	2.74e-12
LPG Production	Benzo[a]anthracene	secondary	3.74e-15	2.65e-12

 Table M-18. LCD LCIA Results for the Photochemical Smog Impact Category

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
LPG Production	Chrysene	secondary	3.59e-15	2.54e-12
LPG Production	Indeno(1,2,3-cd)pyrene	secondary	2.78e-15	1.97e-12
LPG Production	Benzo[b,j,k]fluoranthene	secondary	2.21e-15	1.56e-12
LPG Production	Benzo[a]pyrene	secondary	2.05e-15	1.45e-12
LPG Production	Benzo[g,h,i]perylene	secondary	1.66e-15	1.18e-12
LPG Production	5-Methyl chrysene	secondary	4.41e-16	3.12e-13
LPG Production	Polycyclic aromatic hydrocarbons	secondary	1.75e-16	1.24e-13
Natural Gas Prod.	Polycyclic aromatic hydrocarbons	secondary	-2.97e-16	-2.10e-13
LCD incineration	Polycyclic aromatic hydrocarbons	secondary	-4.52e-15	-3.20e-12
Natural Gas Prod.	5-Methyl chrysene	secondary	-6.43e-14	-4.55e-11
Fuel Oil #4 Prod.	Polycyclic aromatic hydrocarbons	secondary	-7.19e-14	-5.08e-11
Fuel Oil #4 Prod.	5-Methyl chrysene	secondary	-2.66e-13	-1.88e-10
Natural Gas Prod.	Aromatic hydrocarbons	secondary	-2.67e-13	-1.89e-10
Natural Gas Prod.	Benzo[b,j,k]fluoranthene	secondary	-3.22e-13	-2.27e-10
Natural Gas Prod.	2-Chloroacetophenone	secondary	-5.65e-13	-3.99e-10
Natural Gas Prod.	Benzo[a]pyrene	secondary	-7.93e-13	-5.61e-10
Natural Gas Prod.	Benzo[g,h,i]perylene	secondary	-8.04e-13	-5.69e-10
Natural Gas Prod.	2,4-Dinitrotoluene	secondary	-8.19e-13	-5.79e-10
Fuel Oil #4 Prod.	Benzo[g,h,i]perylene	secondary	-8.74e-13	-6.18e-10
Fuel Oil #4 Prod.	Benzo[a]pyrene	secondary	-1.18e-12	-8.37e-10
Natural Gas Prod.	Indeno(1,2,3-cd)pyrene	secondary	-1.24e-12	-8.78e-10
Fuel Oil #4 Prod.	Benzo[b,j,k]fluoranthene	secondary	-1.33e-12	-9.40e-10
Natural Gas Prod.	Chrysene	secondary	-1.36e-12	-9.62e-10
Natural Gas Prod.	Benzo[a]anthracene	secondary	-1.41e-12	-9.98e-10
Fuel Oil #4 Prod.	Indeno(1,2,3-cd)pyrene	secondary	-1.51e-12	-1.07e-09
Natural Gas Prod.	1,1,1-Trichloroethane	secondary	-1.61e-12	-1.14e-09
Natural Gas Prod.	Acenaphthylene	secondary	-1.76e-12	-1.25e-09
Natural Gas Prod.	Chlorobenzene	secondary	-1.78e-12	-1.25e-09
Natural Gas Prod.	Ethylene dibromide	secondary	-1.83e-12	-1.30e-09
Fuel Oil #4 Prod.	Benzo[a]anthracene	secondary	-1.98e-12	-1.40e-09
Fuel Oil #4 Prod.	Chrysene	secondary	-1.99e-12	-1.41e-09
Natural Gas Prod.	Anthracene	secondary	-2.02e-12	-1.43e-09
Fuel Oil #4 Prod.	2-Chloroacetophenone	secondary	-2.34e-12	-1.65e-09
Natural Gas Prod.	1.2-Dichloroethane	secondary	-3.23e-12	-2.28e-09
Natural Gas Prod	Acenaphthene	secondary	-3 34e-12	-2.36e-09
Fuel Oil #4 Prod.	2.4-Dinitrotoluene	secondary	-3.39e-12	-2.39e-09
Natural Gas Prod	Ethyl Chloride	secondary	-3 39e-12	-2.40e-09
Natural Gas Prod	Tetrachloroethylene	secondary	-3 47e-12	-2 45e-09
Fuel Oil #4 Prod	Anthracene	secondary	-3 55e-12	-2 51e-09
Fuel Oil #4 Prod	Acenaphthylene	secondary	-3 73e-12	-2 64e-09
Natural Gas Prod	Fluoranthene	secondary	-3.87e-12	-2 74e-09
Natural Gas Prod	Pyrene	secondary	-3.89e-12	-2 75e-09
I CD incineration	Aromatic hydrocarbons	secondary	-4 07e-12	-2.88e-09
LCD incineration	2-Methylnanhthalene	secondary	-4.30e-12	-2.00C-07
Natural Gas Prod	Eluorene	secondary	-4.34e-12	-3.07e-09
Natural Gas Prod	Chloroform	secondary	-4.76e 12	-3 37e-09
Natural Gas Prod	Binhanyl	secondary	-4.700-12	3 510 00
I CD incineration	Dichlorohenzene (mixed isomers)	secondary	-4.9/0-12	-4.100.00
Evel Oil #4 Prod	Dyrana	secondary	-5.950-12	4 200 00
Fuel Oli #4 Prou.	r yiche	secondary	-0.000-12	-4.290-09
Inatural Gas Prod.	v myr acetate	secondary	-0.51e-12	-4.00e-09

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
Fuel Oil #4 Prod.	1,1,1-Trichloroethane	secondary	-6.67e-12	-4.72e-09
Fuel Oil #4 Prod.	Chlorobenzene	secondary	-7.34e-12	-5.19e-09
Fuel Oil #4 Prod.	Ethylene dibromide	secondary	-7.59e-12	-5.37e-09
Fuel Oil #4 Prod.	Acenaphthene	secondary	-8.56e-12	-6.05e-09
Fuel Oil #4 Prod.	2-Methylnaphthalene	secondary	-9.12e-12	-6.45e-09
Fuel Oil #4 Prod.	Fluoranthene	secondary	-9.92e-12	-7.01e-09
Fuel Oil #4 Prod.	Fluorene	secondary	-1.23e-11	-8.66e-09
Fuel Oil #4 Prod.	1,4-Dichlorobenzene	secondary	-1.26e-11	-8.90e-09
Fuel Oil #4 Prod.	1,2-Dichloroethane	secondary	-1.33e-11	-9.44e-09
Natural Gas Prod.	2-Methylnaphthalene	secondary	-1.36e-11	-9.63e-09
Fuel Oil #4 Prod.	Ethyl Chloride	secondary	-1.40e-11	-9.91e-09
Fuel Oil #4 Prod.	Tetrachloroethylene	secondary	-1.43e-11	-1.01e-08
Natural Gas Prod.	Cumene	secondary	-1.55e-11	-1.10e-08
LCD incineration	5-Methyl chrysene	secondary	-1.61e-11	-1.14e-08
Natural Gas Prod.	Phenanthrene	secondary	-1.78e-11	-1.25e-08
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	-1.88e-11	-1.33e-08
Fuel Oil #4 Prod.	Chloroform	secondary	-1.97e-11	-1.39e-08
LCD incineration	Benzo[g,h,i]pervlene	secondary	-2.00e-11	-1.42e-08
Fuel Oil #4 Prod.	Biphenyl	secondary	-2.06e-11	-1.45e-08
Natural Gas Prod.	Dichloromethane	secondary	-2.34e-11	-1.65e-08
Fuel Oil #4 Prod.	Vinvl acetate	secondary	-2.69e-11	-1.90e-08
LCD incineration	Benzo[a]pyrene	secondary	-2.81e-11	-1.99e-08
Fuel Oil #4 Prod.	Phenanthrene	secondary	-3.95e-11	-2.80e-08
Natural Gas Prod.	Methyl chloride	secondary	-4.28e-11	-3.02e-08
Natural Gas Prod.	Acetophenone	secondary	-4.39e-11	-3.10e-08
LCD incineration	Indeno(1.2.3-cd)pyrene	secondary	-4.51e-11	-3.19e-08
Natural Gas Prod.	Phenol	secondary	-4.68e-11	-3.31e-08
Natural Gas Prod.	Methyl tert-butyl ether	secondary	-5.35e-11	-3.78e-08
Natural Gas Prod.	Benzyl chloride	secondary	-5.65e-11	-3.99e-08
LCD incineration	Benzo[a]anthracene	secondary	-5.90e-11	-4.17e-08
Fuel Oil #4 Prod.	Cumene	secondary	-6.41e-11	-4.53e-08
Fuel Oil #4 Prod	Aromatic hydrocarbons	secondary	-6 47e-11	-4 57e-08
Natural Gas Prod.	Styrene	secondary	-7.31e-11	-5.17e-08
LCD incineration	Chrysene	secondary	-7 36e-11	-5.21e-08
LCD incineration	Benzolb i klfluoranthene	secondary	-8.06e-11	-5 70e-08
Fuel Oil #4 Prod	Dichloromethane	secondary	-9 68e-11	-6 84e-08
LCD incineration	Anthracene	secondary	-1 54e-10	-1 09e-07
Fuel Oil #4 Prod	Methyl chloride	secondary	-1 77e-10	-1 25e-07
Fuel Oil #4 Prod	Acetophenone	secondary	-1.81e-10	-1 28e-07
I CD incineration	A cenaphthylene	secondary	-1 84e-10	-1 30e-07
Fuel Oil #4 Prod	Phenol	secondary	-1 93e-10	-1 37e-07
I CD incineration	2 4-Dinitrotoluene	secondary	-2 05e-10	-1 45e-07
Fuel Oil #4 Prod	Methyl tert-butyl ether	secondary	-2.050 10	-1 56e-07
Fuel Oil #4 Prod	Benzyl chloride	secondary	-2.210-10	-1.65e-07
LCD incineration	Pyrene	secondary	-2.540-10 -2.43e-10	-1 72e-07
Natural Gas Prod	Bromomethane	secondary	-2.45e 10	-1 73e-07
Natural Gas Prod	Fthylbenzene	secondary	-2.430-10 _2.78e 10	-1 97e-07
Fuel Oil #4 Prod	Styrene	secondary	-2.780-10 -3.02e-10	-2 14e-07
1 401 011 // 1100.	Styrono	secondary	5.020-10	2.140 07

 Table M-18.
 LCD LCIA Results for the Photochemical Smog Impact Category

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
LCD incineration	Acenaphthene	secondary	-3.74e-10	-2.65e-07
LCD incineration	1,1,1-Trichloroethane	secondary	-4.05e-10	-2.86e-07
Natural Gas Prod.	n-Propane	secondary	-4.43e-10	-3.14e-07
Natural Gas Prod.	Naphthalene	secondary	-4.44e-10	-3.14e-07
LCD incineration	Chlorobenzene	secondary	-4.45e-10	-3.15e-07
LCD incineration	Ethylene dibromide	secondary	-4.60e-10	-3.25e-07
Natural Gas Prod.	Acrolein	secondary	-4.94e-10	-3.49e-07
LCD incineration	Fluoranthene	secondary	-5.21e-10	-3.68e-07
Fuel Oil #4 Prod.	Naphthalene	secondary	-5.45e-10	-3.85e-07
LCD incineration	Fluorene	secondary	-6.68e-10	-4.72e-07
Natural Gas Prod.	Methyl ethyl ketone	secondary	-7.09e-10	-5.01e-07
LCD incineration	1,2-Dichloroethane	secondary	-8.04e-10	-5.68e-07
LCD incineration	Ethyl Chloride	secondary	-8.50e-10	-6.01e-07
LCD incineration	Tetrachloroethylene	secondary	-8.65e-10	-6.11e-07
Natural Gas Prod.	Propionaldehyde	secondary	-8.80e-10	-6.22e-07
Natural Gas Prod.	Acetaldehyde	secondary	-9.70e-10	-6.86e-07
Fuel Oil #4 Prod.	Bromomethane	secondary	-1.01e-09	-7.15e-07
Fuel Oil #4 Prod.	Ethylbenzene	secondary	-1.15e-09	-8.10e-07
LCD incineration	Chloroform	secondary	-1.19e-09	-8.40e-07
LCD incineration	Biphenyl	secondary	-1.25e-09	-8.81e-07
LCD incineration	Vinyl acetate	secondary	-1.63e-09	-1.15e-06
Fuel Oil #4 Prod.	o-xylene	secondary	-1.63e-09	-1.16e-06
Fuel Oil #4 Prod.	n-Propane	secondary	-1.66e-09	-1.17e-06
Natural Gas Prod.	Isophorone	secondary	-1.70e-09	-1.20e-06
LCD incineration	Phenanthrene	secondary	-1.98e-09	-1.40e-06
Fuel Oil #4 Prod.	Acrolein	secondary	-2.04e-09	-1.44e-06
Fuel Oil #4 Prod.	Methyl ethyl ketone	secondary	-2.93e-09	-2.07e-06
Fuel Oil #4 Prod.	Propionaldehyde	secondary	-3.64e-09	-2.57e-06
LCD incineration	Cumene	secondary	-3.89e-09	-2.75e-06
Fuel Oil #4 Prod.	Acetaldehyde	secondary	-4.01e-09	-2.84e-06
Fuel Oil #4 Prod.	Toluene	secondary	-4.52e-09	-3.20e-06
LCD incineration	Dichloromethane	secondary	-5.86e-09	-4.14e-06
Fuel Oil #4 Prod.	Isophorone	secondary	-7.01e-09	-4.96e-06
Natural Gas Prod.	Toluene	secondary	-9.63e-09	-6.80e-06
LCD incineration	Methyl chloride	secondary	-1.07e-08	-7.58e-06
LCD incineration	Acetophenone	secondary	-1.10e-08	-7.77e-06
LCD incineration	Phenol	secondary	-1.17e-08	-8.29e-06
LCD incineration	Methyl tert-butyl ether	secondary	-1.34e-08	-9.49e-06
LCD incineration	Naphthalene	secondary	-1.35e-08	-9.57e-06
LCD incineration	Benzyl chloride	secondary	-1.42e-08	-1.00e-05
Natural Gas Prod.	o-xylene	secondary	-1.74e-08	-1.23e-05
LCD incineration	Styrene	secondary	-1.83e-08	-1.30e-05
LCD incineration	Xylene (mixed isomers)	secondary	-2.54e-08	-1.79e-05
Natural Gas Prod.	Formaldehyde	secondary	-3.37e-08	-2.38e-05
LCD incineration	Bromomethane	secondary	-6.14e-08	-4.34e-05
LCD incineration	Ethylbenzene	secondary	-6.84e-08	-4.84e-05
LCD incineration	n-Propane	secondary	-9.86e-08	-6.97e-05
LCD incineration	Acrolein	secondary	-1.24e-07	-8.75e-05
LCD incineration	Toluene	secondary	-1.59e-07	-1.13e-04
LCD incineration	Methyl ethyl ketone	secondary	-1.78e-07	-1.26e-04

Process Group	Material	LCI Data Type	Photo. Smog (kg ethene- equivalents)	% of Total
LCD incineration	Butane	secondary	-1.97e-07	-1.39e-04
LCD incineration	Hexane	secondary	-2.03e-07	-1.44e-04
LCD incineration	Propionaldehyde	secondary	-2.21e-07	-1.56e-04
LCD incineration	Acetaldehyde	secondary	-2.43e-07	-1.72e-04
LCD incineration	Pentane	secondary	-2.50e-07	-1.76e-04
LCD incineration	Ethane	secondary	-2.90e-07	-2.05e-04
Fuel Oil #4 Prod.	Hexane	secondary	-3.74e-07	-2.65e-04
Fuel Oil #4 Prod.	Butane	secondary	-4.17e-07	-2.95e-04
LCD incineration	Isophorone	secondary	-4.25e-07	-3.01e-04
Fuel Oil #4 Prod.	Pentane	secondary	-5.30e-07	-3.74e-04
Natural Gas Prod.	Hexane	secondary	-5.59e-07	-3.95e-04
Fuel Oil #4 Prod.	Ethane	secondary	-6.16e-07	-4.36e-04
Natural Gas Prod.	Butane	secondary	-6.24e-07	-4.41e-04
Natural Gas Prod.	Pentane	secondary	-7.91e-07	-5.60e-04
Natural Gas Prod.	Ethane	secondary	-9.20e-07	-6.51e-04
LCD incineration	Formaldehyde	secondary	-1.06e-06	-7.52e-04
Fuel Oil #4 Prod.	Aldehydes	secondary	-1.32e-06	-9.32e-04
Fuel Oil #4 Prod.	Formaldehyde	secondary	-1.34e-06	-9.48e-04
Natural Gas Prod.	Aldehydes	secondary	-1.58e-06	-1.12e-03
Fuel Oil #4 Prod.	Benzene	secondary	-5.41e-06	-3.82e-03
LCD incineration	Benzene	secondary	-6.58e-06	-4.65e-03
Natural Gas Prod.	Hydrocarbons, remaining unspeciated	secondary	-8.20e-06	-5.80e-03
Fuel Oil #4 Prod.	Methane	secondary	-1.30e-05	-9.16e-03
LCD incineration	Aldehydes	secondary	-4.38e-05	-3.10e-02
LCD incineration	Methane	secondary	-6.89e-05	-4.87e-02
Natural Gas Prod.	Benzene	secondary	-7.92e-05	-5.60e-02
LCD incineration	Nonmethane hydrocarbons, remaining unspeciated	secondary	-9.99e-05	-7.07e-02
Fuel Oil #4 Prod.	Nonmethane hydrocarbons, remaining unspeciated	secondary	-1.03e-04	-7.27e-02
Natural Gas Prod.	Methane	secondary	-1.16e-04	-8.22e-02
LCD incineration	Hydrocarbons, remaining unspeciated	secondary	-1.24e-04	-8.77e-02
Fuel Oil #4 Prod.	Hydrocarbons, remaining unspeciated	secondary	-1.53e-04	-1.08e-01
Natural Gas Prod.	Nonmethane hydrocarbons, remaining unspeciated	secondary	-2.98e-04	-2.11e-01
Total End-of-life			-1.11e-03	-7.82e-01
Total All Life-cycle Stages			1.41e-01	1.00e+02

 Table M-18.
 LCD LCIA Results for the Photochemical Smog Impact Category

				9/ CT / 1
Process Group	Material	LCI Data Type	Acidification (kg SO2 Equivalents)	% of Total
Materials Processing Life-cycle Stage				
Invar	Sulfur dioxide	secondary	2.50e-01	4.75e+00
Steel Prod., cold-rolled, semi-finished	Sulfur dioxide	secondary	3.84e-02	7.31e-01
Aluminum Prod.	Sulfur dioxide	secondary	2.09e-02	3.98e-01
Polycarbonate Production	Nitrogen dioxide	secondary	1.36e-02	2.59e-01
Polycarbonate Production	Sulfur dioxide	secondary	1.20e-02	2.29e-01
Steel Prod., cold-rolled, semi-finished	Nitrogen dioxide	secondary	9.36e-03	1.78e-01
Lead	Sulfur dioxide	secondary	6.92e-03	1.32e-01
Styrene-butadiene Copolymer Prod.	Sulfur oxides	secondary	5.71e-03	1.09e-01
Aluminum Prod.	Nitrogen dioxide	secondary	4.99e-03	9.51e-02
Styrene-butadiene Copolymer Prod.	Nitrogen oxides	secondary	4.89e-03	9.32e-02
ABS Production	Sulfur dioxide	secondary	4.24e-03	8.07e-02
Ferrite mfg.	Sulfur dioxide	secondary	3.56e-03	6.78e-02
Invar	Nitrogen dioxide	secondary	3.55e-03	6.75e-02
ABS Production	Nitrogen dioxide	secondary	3.26e-03	6.21e-02
Lead	Nitrogen dioxide	secondary	2.93e-03	5.58e-02
Polystyrene Prod., high-impact	Sulfur dioxide	secondary	1.81e-03	3.45e-02
Ferrite mfg.	Nitrogen dioxide	secondary	1.42e-03	2.70e-02
Polystyrene Prod., high-impact	Nitrogen dioxide	secondary	1.27e-03	2.42e-02
Steel Prod cold-rolled semi-finished	Ammonia	secondary	7 79e-04	1 48e-02
ABS Production	Hydrochloric acid	secondary	5 63e-04	1.07e-02
Steel Prod cold-rolled semi-finished	Nitrous oxide	secondary	5 44e-04	1.04e-02
Steel Prod. cold-rolled semi-finished	Hydrochloric acid	secondary	3 93e-04	7 48e-03
ABS Production	Hydrofluoric acid	secondary	3.90e-04	7.42e-03
Aluminum Prod	Hydrofluoric acid	secondary	3.85e-04	7.12e 03
Invar	Hydrochloric acid	secondary	3 56e-04	6 77e-03
Aluminum Prod	Hydrochloric acid	secondary	3.43e-04	6.53e-03
Lead	Hydrochloric acid	secondary	2.06e-04	3.92e-03
Ferrite mfg	Hydrochloric acid	secondary	1.33e-04	2.54e-03
Aluminum Prod	Nitrous oxide	secondary	9.67e-05	1.84e-03
Polycarbonate Production	Hydrochloric acid	secondary	9.07e 05 8.94e-05	1.04e 03
Inver	Hydrofluoric acid	secondary	4 30e 05	8 10e 04
Steel Prod cold-rolled semi-finished	Hydrofluoric acid	secondary	4.50e-05	5.77e-04
Steel Prod. cold rolled semi finished	Hydrogen sulfide	secondary	2 55e 05	4 870 04
Lend	Hydrofluoric acid	secondary	2.556-05	4.16e.04
ABS Production	Hydrogen sulfide	secondary	2.180-05	3 620 04
Aluminum Prod	Ammonia	secondary	1.900-05	3 22e 04
Lead	Nitrous ovida	secondary	1.090-05	2 710 04
Leau	Undragen gulfide	secondary	1.426-05	2.716-04
Styrana hutadiana Canalymar Drad	Hydroghlaria acid	secondary	1.376-03	2.000-04
Lead	Ammonio	secondary	1.316-03	2.306-04
Lead		secondary	1.500-05	2.476-04
Ferrite mfg.	Hydroniuonic acid	secondary	0.722.06	1.946-04
Ferrite mig.	Nitrana anida	secondary	9.728-00	1.836-04
Ferrite mig.	Nurous oxide	secondary	7.446-06	1.42e-04
Aluminum Prod.	Hydrogen suifide	secondary	/.08e-06	1.35e-04
IIIvar Delevente predention	Ammonia	secondary	6.66e-06	1.2/e-04
Polycarbonate Production	Hydrofluoric acid	secondary	5.91e-06	1.13e-04
	Hydrogen sulfide	secondary	5.57e-06	1.06e-04
Polystyrene Prod., nigh-impact	Hydrochloric acid	secondary	4.66e-06	8.8/e-05
Polycarbonate Production	Hydrogen sulfide	secondary	3.47e-06	6.61e-05
Ferrite mfg.	Ammonia	secondary	2.20e-06	4.19e-05

Table M-19. CRT LCIA Results for the Acidification Impact Categor

Process Group	Material	LCI Data Type	Acidification (kg SO2 Equivalents)	% of Total
ABS Production	Ammonia	secondary	1.59e-06	3.03e-05
Styrene-butadiene Copolymer Prod.	Hydrofluoric acid	secondary	6.62e-07	1.26e-05
Styrene-butadiene Copolymer Prod.	Ammonia	secondary	3.89e-07	7.41e-06
Styrene-butadiene Copolymer Prod.	Hydrogen sulfide	secondary	3.89e-07	7.41e-06
Polycarbonate Production	Nitrous oxide	secondary	3.23e-07	6.15e-06
Polycarbonate Production	Sulfuric acid	secondary	3.00e-07	5.72e-06
ABS Production	Nitrous oxide	secondary	1.48e-07	2.82e-06
Styrene-butadiene Copolymer Prod.	Nitrous oxide	secondary	1.45e-07	2.76e-06
ABS Production	Sulfuric acid	secondary	1.38e-07	2.62e-06
Styrene-butadiene Copolymer Prod.	Sulfuric acid	secondary	1.34e-07	2.56e-06
Invar	Nitrous oxide	secondary	3.39e-08	6.46e-07
Invar	Sulfuric acid	secondary	5.66e-11	1.08e-09
Ferrite mfg.	Sulfuric acid	secondary	5.53e-11	1.05e-09
Total Materials Processing			3.93e-01	7.48e+00
Manufacturing Life-cycle Stage				
LPG Production	Sulfur oxides	secondary	8.01e-01	1.53e+01
LPG Production	Nitrogen oxides	secondary	4.01e-01	7.64e+00
Japanese Electric Grid	Sulfur dioxide	model/secondary	8.62e-02	1.64e+00
US electric grid	Sulfur dioxide	model/secondary	3.98e-02	7.58e-01
Glass/frit	Nitrogen oxides	primary	3.09e-02	5.88e-01
Japanese Electric Grid	Nitrogen oxides	model/secondary	2.85e-02	5.42e-01
US electric grid	Nitrogen oxides	model/secondary	1.32e-02	2.50e-01
LPG Production	Nitrous oxide	secondary	1.14e-02	2.16e-01
CRT tube mfg.	Sulfur oxides	primary	9.96e-03	1.90e-01
LPG Production	Hydrochloric acid	secondary	7.07e-03	1.35e-01
Natural Gas Prod.	Nitrogen oxides	secondary	6.53e-03	1.24e-01
LPG Production	Hydrogen sulfide	secondary	5.76e-03	1.10e-01
Fuel Oil #6 Prod.	Sulfur oxides	secondary	5.47e-03	1.04e-01
LPG Production	Ammonia	secondary	4.09e-03	7.79e-02
Fuel Oil #6 Prod.	Nitrogen oxides	secondary	2.87e-03	5.46e-02
Fuel Oil #2 Prod.	Sulfur oxides	secondary	2.44e-03	4.64e-02
LPG Production	Hydrofluoric acid	secondary	1.61e-03	3.06e-02
CRT tube mfg.	Nitrogen oxides	primary	1.52e-03	2.89e-02
US electric grid	Hydrochloric acid	model/secondary	1.51e-03	2.88e-02
Fuel Oil #2 Prod.	Nitrogen oxides	secondary	1.23e-03	2.34e-02
Japanese Electric Grid	Hydrochloric acid	model/secondary	1.21e-03	2.30e-02
Glass/frit	Nitrogen oxides	primary	3.73e-04	7.11e-03
US electric grid	Hydrofluoric acid	model/secondary	3.44e-04	6.54e-03
Natural Gas Prod.	Ammonia	secondary	2.94e-04	5.61e-03
Natural Gas Prod.	Sulfur oxides	secondary	2.76e-04	5.25e-03
Japanese Electric Grid	Hydrofluoric acid	model/secondary	2.74e-04	5.22e-03
Fuel Oil #4 Prod.	Sulfur oxides	secondary	2.45e-04	4.66e-03
Fuel Oil #4 Prod.	Nitrogen oxides	secondary	1.26e-04	2.39e-03
Fuel Oil #6 Prod.	Nitrous oxide	secondary	1.02e-04	1.93e-03
Japanese Electric Grid	Nitrous oxide	model/secondary	7.82e-05	1.49e-03
Fuel Oil #6 Prod.	Hydrogen sulfide	secondary	5.97e-05	1.14e-03
Fuel Oil #6 Prod.	Hydrochloric acid	secondary	5.70e-05	1.09e-03
Glass/frit	Sulfur oxides	primary	5.08e-05	9.68e-04
US electric grid	Nitrous oxide	model/secondary	3.80e-05	7.23e-04

Process Group	Material	LCI Data Type	Acidification (kg SO2 Equivalents)	% of Total
Fuel Oil #2 Prod.	Nitrous oxide	secondary	3.63e-0	5 6.91e-04
Fuel Oil #2 Prod.	Hydrochloric acid	secondary	2.22e-0	5 4.22e-04
Fuel Oil #6 Prod.	Ammonia	secondary	2.20e-0	5 4.20e-04
Fuel Oil #2 Prod	Hydrogen sulfide	secondary	1 90e-0	3 62e-04
Fuel Oil #6 Prod.	Hydrofluoric acid	secondary	1.30e-0	5 2.47e-04
Natural Gas Prod	Hydrochloric acid	secondary	1.23e-0	5 2.34e-04
Fuel Oil #2 Prod.	Ammonia	secondary	1.21e-0	5 2.30e-04
Natural Gas Prod	Nitrous oxide	secondary	8 08e-00	5 1 54e-04
Fuel Oil #2 Prod.	Hydrofluoric acid	secondary	5.04e-00	5 9.60e-05
Fuel Oil #4 Prod	Nitrous oxide	secondary	4 02e-00	5 7.66e-05
Natural Gas Prod	Hydrofluoric acid	secondary	2.79e-00	5 531e-05
Fuel Oil #4 Prod	Hydrochloric acid	secondary	2.36e-00	5 4 50e-05
Fuel Oil #4 Prod	Hydrogen sulfide	secondary	2 23e-00	5 4 24e-05
Fuel Oil #4 Prod	Ammonia	secondary	1 12e-00	5 2 13e-05
Fuel Oil #4 Prod	Hydrofluoric acid	secondary	5 37e-0	1 02e-05
Natural Gas Prod	Hydrogen sulfide	secondary	1 2.7e-0	7 2.41e-06
Total Manufacturing	i i jai ogen sannae	secondary	1.47.0	2.70 ± 0.1
Use. Maintenance and Renair Life-cycle	Stage		1.4/0+00	2./9e+01
US electric grid	Sulfur dioxide	model/secondary	2 49e+00) $4.75e+01$
US electric grid	Nitrogen oxides	model/secondary	8 24e-0	1.57e+01
US electric grid	Hydrochloric acid	model/secondary	9.47e-0	1.370 ± 01
US electric grid	Hydrofluoric acid	model/secondary	2.15e-02	2 4 10e-01
US electric grid	Nitrous oxide	model/secondary	2.13e 01 2.38e-01	4 53e-02
Total Use Maintanance and Renair	I vidous oxide	model/secondary	2.560-0.	6.5401.01
Find of life Life could Stage			5.440+00	0.54e+01
End-of-life Life-cycle Stage				
CRT landfilling	Nitrogen dioxide	primary	1.30e-03	2.47e-02
CRT Incineration	Sulfur dioxide	secondary	3.00e-04	5.72e-03
CRT landfilling	Sulfur dioxide	primary	2.80e-04	5.33e-03
US electric grid	Sulfur dioxide	model/secondary	2.49e-04	4.75e-03
US electric grid	Nitrogen oxides	model/secondary	8.24e-03	5 1.57e-03
US electric grid	Hydrochloric acid	model/secondary	9.48e-00	5 1.80e-04
LPG Production	Sulfur oxides	secondary	6.92e-00	1.32e-04
LPG Production	Nitrogen oxides	secondary	3.46e-00	6.59e-05
US electric grid	Hydrofluoric acid	model/secondary	2.15e-00	4.10e-05
CRT landfilling	Hydrochloric acid	primary	9.00e-0	1.71e-05
CRT landfilling	Hydrogen sulfide	primary	4.33e-0	8.25e-06
US electric grid	Nitrous oxide	model/secondary	2.38e-0	4.53e-06
LPG Production	Nitrous oxide	secondary	9.80e-08	1.87e-06
LPG Production	Hydrochloric acid	secondary	6.10e-08	1.16e-06
LPG Production	Hydrogen sulfide	secondary	4.97e-08	9.47e-07
LPG Production	Ammonia	secondary	3.53e-08	6.73e-07
LPG Production	Hydrofluoric acid	secondary	1.39e-08	2.64e-07
Natural Gas Prod.	Hydrogen sulfide	secondary	-6.34e-08	-1.21e-06
CRT Incineration	Hydrogen sulfide	secondary	-1.15e-00	-2.19e-05
Natural Gas Prod.	Hydrofluoric acid	secondary	-1.40e-00	-2.66e-05
Natural Gas Prod.	Nitrous oxide	secondary	-4.05e-00	-7.72e-05
Fuel Oil #4 Prod.	Hydrofluoric acid	secondary	-5.78e-00	-1.10e-04
Natural Gas Prod.	Hydrochloric acid	secondary	-6.16e-00	-1.17e-04
Fuel Oil #4 Prod.	Ammonia	secondary	-1.20e-05	-2.29e-04
Fuel Oil #4 Prod.	Hydrogen sulfide	secondary	-2.40e-03	-4.56e-04

			F	
Process Group	Material	LCI Data Type	Acidification (kg SO2 Equivalents)	% of Total
Fuel Oil #4 Prod.	Hydrochloric acid	secondary	-2.54e-05	-4.84e-04
CRT Incineration	Ammonia	secondary	-3.91e-05	-7.44e-04
Fuel Oil #4 Prod.	Nitrous oxide	secondary	-4.33e-05	-8.24e-04
Natural Gas Prod.	Sulfur oxides	secondary	-1.38e-04	-2.63e-03
Natural Gas Prod.	Ammonia	secondary	-1.48e-04	-2.81e-03
CRT Incineration	Hydrofluoric acid	secondary	-3.57e-04	-6.80e-03
CRT Incineration	Nitrous oxide	secondary	-7.29e-04	-1.39e-02
CRT Incineration	Hydrochloric acid	secondary	-8.95e-04	-1.70e-02
Fuel Oil #4 Prod.	Nitrogen oxides	secondary	-1.35e-03	-2.57e-02
Fuel Oil #4 Prod.	Sulfur oxides	secondary	-2.63e-03	-5.01e-02
Natural Gas Prod.	Nitrogen oxides	secondary	-3.27e-03	-6.23e-02
CRT Incineration	Nitrogen oxides	secondary	-8.78e-03	-1.67e-01
CRT Incineration	Sulfur oxides	secondary	-2.70e-02	-5.14e-01
Total End-of-life			-4.32e-02	-8.23e-01
Total All Life-cycle Stages			5.25e+00	1.00e+02

Table M-19.	CRT LCIA	Results for th	e Acidification	Impact (Category
-------------	----------	-----------------------	-----------------	----------	----------

Process Group	Material	LCI Data Type	Acidification (kg SO2 Equivalents)	% of Total
Materials Processing Life-cycle Stage				
Natural Gas Prod.	Nitrogen oxides	secondary	4.56e-01	1.54e+01
Natural Gas Prod.	Ammonia	secondary	2.06e-02	6.95e-01
Natural Gas Prod.	Sulfur oxides	secondary	1.93e-02	6.51e-01
Steel Prod., cold-rolled, semi-finished	Sulfur dioxide	secondary	1.88e-02	6.36e-01
PMMA Sheet Prod.	Sulfur dioxide	secondary	1.30e-02	4.41e-01
Aluminum Prod.	Sulfur dioxide	secondary	7.77e-03	2.63e-01
Polycarbonate Production	Nitrogen dioxide	secondary	7.58e-03	2.56e-01
PMMA Sheet Prod.	Nitrogen dioxide	secondary	7.52e-03	2.54e-01
Polycarbonate Production	Sulfur dioxide	secondary	6.70e-03	2.27e-01
Steel Prod., cold-rolled, semi-finished	Nitrogen dioxide	secondary	4.59e-03	1.55e-01
PET Resin Production	Sulfur oxides	secondary	3.99e-03	1.35e-01
Styrene-butadiene Copolymer Prod.	Sulfur oxides	secondary	2.50e-03	8.43e-02
Styrene-butadiene Copolymer Prod.	Nitrogen oxides	secondary	2.14e-03	7.23e-02
Aluminum Prod.	Nitrogen dioxide	secondary	1.86e-03	6.28e-02
PET Resin Production	Nitrogen oxides	secondary	1.21e-03	4.08e-02
Natural Gas Prod.	Hydrochloric acid	secondary	8.57e-04	2.90e-02
Natural Gas Prod.	Nitrous oxide	secondary	5.65e-04	1.91e-02
Steel Prod., cold-rolled, semi-finished	Ammonia	secondary	3.82e-04	1.29e-02
Steel Prod., cold-rolled, semi-finished	Nitrous oxide	secondary	2.67e-04	9.01e-03
Natural Gas Prod.	Hydrofluoric acid	secondary	1.95e-04	6.58e-03
Steel Prod., cold-rolled, semi-finished	Hydrochloric acid	secondary	1.93e-04	6.51e-03
Aluminum Prod.	Hydrofluoric acid	secondary	1.43e-04	4.84e-03
Aluminum Prod.	Hydrochloric acid	secondary	1.28e-04	4.31e-03
PMMA Sheet Prod.	Hydrochloric acid	secondary	5.74e-05	1.94e-03
Polycarbonate Production	Hydrochloric acid	secondary	4.99e-05	1.69e-03
Aluminum Prod.	Nitrous oxide	secondary	3.60e-05	1.22e-03
PET Resin Production	Hydrochloric acid	secondary	2.80e-05	9.45e-04
PMMA Sheet Prod.	Ammonia	secondary	1.51e-05	5.12e-04
Steel Prod., cold-rolled, semi-finished	Hydrofluoric acid	secondary	1.48e-05	5.02e-04
Steel Prod., cold-rolled, semi-finished	Hydrogen sulfide	secondary	1.25e-05	4.23e-04
Natural Gas Prod.	Hydrogen sulfide	secondary	8.84e-06	2.99e-04
Aluminum Prod.	Ammonia	secondary	6.30e-06	2.13e-04
Styrene-butadiene Copolymer Prod.	Hydrochloric acid	secondary	5.73e-06	1.94e-04
PMMA Sheet Prod.	Hydrofluoric acid	secondary	5.52e-06	1.87e-04
Polycarbonate Production	Hydrofluoric acid	secondary	3.30e-06	1.12e-04
PMMA Sheet Prod.	Hydrogen sulfide	secondary	2.88e-06	9.75e-05
Aluminum Prod.	Hydrogen sulfide	secondary	2.64e-06	8.91e-05
Polycarbonate Production	Hydrogen sulfide	secondary	1.94e-06	6.55e-05
PET Resin Production	Hydrofluoric acid	secondary	1.45e-06	4.91e-05
Styrene-butadiene Conolymer Prod	Hydrofluoric acid	secondary	2.89e-07	9 78e-06
Polycarbonate Production	Nitrous oxide	secondary	1.81e-07	6.10e-06
PET Resin Production	Ammonia	secondary	1.71e-07	5.77e-06
PET Resin Production	Hydrogen sulfide	secondary	1.71e-07	5 77e-06
Styrene-butadiene Conolymer Prod	Ammonia	secondary	1.70e-07	5 75e-06
Styrene-butadiene Copolymer Prod.	Hydrogen sulfide	secondary	1.70e-07	5.75e-06
Polycarbonate Production	Sulfuric acid	secondary	1 68e-07	5.66e-06
PMMA Sheet Prod.	Nitrous oxide	secondary	1 34e-07	4 54e-06
PMMA Sheet Prod.	Sulfuric acid	secondary	1.540 07 1.25e-07	4 21e-06
PET Resin Production	Nitrous oxide	secondary	6 35e-08	2.15e-06
Styrene-butadiene Copolymer Prod	Nitrous oxide	secondary	6 33e-08	2.150 00 2.14e-06
Styrene buttatene Coporymer i rou.	THEORS ONLICE	secondary	0.550-00	2.140-00

Process Group	Material	LCI Data Type	Acidification (kg SO2 Equivalents)	% of Total
PET Resin Production	Sulfuric acid	secondary	5.90e-	08 1.99e-06
Styrene-butadiene Copolymer Prod.	Sulfuric acid	secondary	5.88e-	08 1.99e-06
Total Materials Processing			5.76e-	01 1.95e+01
Manufacturing Life-cycle Stage				
Monitor/module	Nitrogen oxides	primary	3.84e-	01 1.30e+01
Japanese Electric Grid	Sulfur dioxide	model/secondary	2.90e-	01 9.80e+00
Monitor/module	Ammonia	primary	1.17e-	01 3.96e+00
Japanese Electric Grid	Nitrogen oxides	model/secondary	9.57e-	02 3.23e+00
Monitor/module	Hydrofluoric acid	primary	8.33e-	02 2.82e+00
Monitor/module	Hydrochloric acid	primary	5.33e-	02 1.80e+00
LPG Production	Sulfur oxides	secondary	3.85e-	02 1.30e+00
Backlight	Nitrogen oxides	primary	2.06e-	02 6.97e-01
LPG Production	Nitrogen oxides	secondary	1.92e-	02 6.50e-01
Natural Gas Prod.	Nitrogen oxides	secondary	1.03e-	02 3.48e-01
US electric grid	Sulfur dioxide	model/secondary	4.83e-	03 1.63e-01
Japanese Electric Grid	Hydrochloric acid	model/secondary	4.06e-	03 1.37e-01
US electric grid	Nitrogen oxides	model/secondary	1.60e-	03 5.39e-02
LCD glass mfg.	Nitrogen oxides	primary	1.43e-	03 4.84e-02
Monitor/module	Sulfur oxides	primary	1.12e-	03 3.77e-02
Japanese Electric Grid	Hydrofluoric acid	model/secondary	9.23e-	04 3.12e-02
LPG Production	Nitrous oxide	secondary	5.45e-	04 1.84e-02
Natural Gas Prod.	Ammonia	secondary	4.64e-	04 1.57e-02
Natural Gas Prod.	Sulfur oxides	secondary	4.35e-	04 1.47e-02
Fuel Oil #4 Prod.	Sulfur oxides	secondary	3.77e-	04 1.28e-02
LPG Production	Hydrochloric acid	secondary	3.39e-	04 1.15e-02
Panel components	Nitrogen oxides	primary	2.88e-	04 9.73e-03
LPG Production	Hydrogen sulfide	secondary	2.77e-	04 9.35e-03
Japanese Electric Grid	Nitrous oxide	model/secondary	2.63e-	04 8.89e-03
LPG Production	Ammonia	secondary	1.96e-	04 6.64e-03
Fuel Oil #4 Prod.	Nitrogen oxides	secondary	1.94e-	04 6.54e-03
Fuel Oil #6 Prod.	Sulfur oxides	secondary	1.86e-	04 6.29e-03
US electric grid	Hydrochloric acid	model/secondary	1.83e-	04 6.20e-03
Monitor/module	Nitric acid	primary	1.37e-	04 4.63e-03
Fuel Oil #2 Prod.	Sulfur oxides	secondary	1.14e-	04 3.84e-03
Fuel Oil #6 Prod.	Nitrogen oxides	secondary	9.76e-	05 3.30e-03
LPG Production	Hydrofluoric acid	secondary	7.71e-	05 2.60e-03
Fuel Oil #2 Prod.	Nitrogen oxides	secondary	5.74e-	05 1.94e-03
Monitor/module	Phosphoric acid	primary	4.75e-	05 1.61e-03
US electric grid	Hydrofluoric acid	model/secondary	4.17e-	05 1.41e-03
Natural Gas Prod.	Hydrochloric acid	secondary	1.94e-	05 6.54e-04
Natural Gas Prod.	Nitrous oxide	secondary	1.27e-	05 4.31e-04
Panel components	Hydrochloric acid	primary	6.44e-	06 2.18e-04
Fuel Oil #4 Prod.	Nitrous oxide	secondary	6.20e-	06 2.10e-04
US electric grid	Nitrous oxide	model/secondary	4.61e-	06 1.56e-04
Natural Gas Prod.	Hydrofluoric acid	secondary	4.40e-	06 1.49e-04
Fuel Oil #4 Prod.	Hydrochloric acid	secondary	3.64e-	06 1.23e-04
Fuel Oil #6 Prod.	Nitrous oxide	secondary	3.46e-	06 1.17e-04
Fuel Oil #4 Prod.	Hydrogen sulfide	secondary	3.43e-	06 1.16e-04
LCD glass mfg.	Sulfur oxides	primary	2.35e-	06 7.96e-05

Process Group	Material	LCI Data Type	Acidification (kg SO2 Equivalents)	% of Total			
Fuel Oil #6 Prod.	Hydrogen sulfide	secondary	2.03e-06	6.87e-05			
Fuel Oil #6 Prod.	Hydrochloric acid	secondary	1.94e-06	6.56e-05			
Fuel Oil #4 Prod.	Ammonia	secondary	1.72e-06	5.83e-05			
Fuel Oil #2 Prod.	Nitrous oxide	secondary	1.69e-06	5.73e-05			
Fuel Oil #2 Prod.	Hydrochloric acid	secondary	1.04e-06	3.50e-05			
Fuel Oil #2 Prod.	Hydrogen sulfide	secondary	8.88e-07	3.00e-05			
Fuel Oil #4 Prod.	Hydrofluoric acid	secondary	8.28e-07	2.80e-05			
Fuel Oil #6 Prod.	Ammonia	secondary	7.50e-07	2.53e-05			
Fuel Oil #2 Prod.	Ammonia	secondary	5.63e-07	1.90e-05			
Fuel Oil #6 Prod.	Hydrofluoric acid	secondary	4.41e-07	1.49e-05			
Fuel Oil #2 Prod.	Hydrofluoric acid	secondary	2.35e-07	7.95e-06			
Natural Gas Prod.	Hydrogen sulfide	secondary	1.99e-07	6.74e-06			
Total Manufacturing			1.13e+00	3.82e+01			
Use, Maintenance and Repair Life	-cvcle Stage						
US electric grid	Sulfur dioxide	model/secondary	9 30e-01	3 14e+01			
US electric grid	Nitrogen oxides	model/secondary	3.08e-01	1.04e+01			
US electric grid	Hydrochloric acid	model/secondary	3 53e-02	1 19e+00			
US electric grid	Hydrofluoric acid	model/secondary	8.03e-03	2 71e-01			
US electric grid	Nitrous oxide	model/secondary	8.88e-04	3.00e-02			
Total Use, Maintenance, and Repa	ir	model becondury	1.28e+00	4.33e+01			
End of life Life avale Store							
L CD landfilling	Nitrogen dioxide	nrimary	3 12e-04	1.06e-02			
US electric grid	Sulfur dioxide	model/secondary	1.77e-04	5.97e-03			
LCD landfilling	Sulfur dioxide	primary	6.74e-05	2 28e-03			
US electric grid	Nitrogen oxides	model/secondary	5.83e-05	1.97e-03			
LCD incineration	Sulfur dioxide	secondary	5.05 e 05	1.77e-03			
US electric grid	Hydrochloric acid	model/secondary	6.71e-06	2.27e-04			
LPG Production	Sulfur oxides	secondary	3 16e-06	1 07e-04			
LPG Production	Nitrogen oxides	secondary	1.58e-06	5.34e-05			
US electric grid	Hydrofluoric acid	model/secondary	1.52e-06	5.15e-05			
LCD landfilling	Hydrochloric acid	primary	2.17e-07	7.33e-06			
US electric grid	Nitrous oxide	model/secondary	1.68e-07	5.69e-06			
LCD landfilling	Hydrogen sulfide	primary	1.04e-07	3.52e-06			
LPG Production	Nitrous oxide	secondary	4.47e-08	1.51e-06			
LPG Production	Hydrochloric acid	secondary	2.78e-08	9.41e-07			
LPG Production	Hydrogen sulfide	secondary	2.27e-08	7.67e-07			
LPG Production	Ammonia	secondary	1.61e-08	5.45e-07			
LPG Production	Hydrofluoric acid	secondary	6.33e-09	2.14e-07			
Natural Gas Prod.	Hydrogen sulfide	secondary	-4.19e-08	-1.42e-06			
Natural Gas Prod.	Hydrofluoric acid	secondary	-9.23e-07	-3.12e-05			
LCD incineration	Hydrogen sulfide	secondary	-9.64e-07	-3.26e-05			
Natural Gas Prod.	Nitrous oxide	secondary	-2.68e-06	-9.04e-05			
Fuel Oil #4 Prod.	Hydrofluoric acid	secondary	-3.81e-06	-1.29e-04			
Natural Gas Prod.	Hydrochloric acid	secondary	-4.06e-06	-1.37e-04			
Fuel Oil #4 Prod.	Ammonia	secondary	-7.95e-06	-2.69e-04			
Fuel Oil #4 Prod.	Hydrogen sulfide	secondary	-1.58e-05	-5.35e-04			
Fuel Oil #4 Prod.	Hydrochloric acid	secondary	-1.68e-05	-5.67e-04			
LCD incineration	Ammonia	secondary	-2.53e-05	-8.56e-04			
Fuel Oil #4 Prod.	Nitrous oxide	secondary	-2.86e-05	-9.66e-04			
Natural Gas Prod.	Sulfur oxides	secondary	-9.13e-05	-3.08e-03			

Process Group	Material	LCI Data Type	Acidification (kg SO2 Equivalents)	% of Total
Natural Gas Prod.	Ammonia	secondary	-9.75e-	05 -3.29e-03
LCD incineration	Hydrofluoric acid	secondary	-2.31e-	04 -7.81e-03
LCD incineration	Nitrous oxide	secondary	-4.72e-	04 -1.60e-02
LCD incineration	Hydrochloric acid	secondary	-5.91e	04 -2.00e-02
Fuel Oil #4 Prod.	Nitrogen oxides	secondary	-8.92e	04 -3.01e-02
Fuel Oil #4 Prod.	Sulfur oxides	secondary	-1.74e-	03 -5.88e-02
Natural Gas Prod.	Nitrogen oxides	secondary	-2.16e-	03 -7.30e-02
LCD incineration	Nitrogen oxides	secondary	-6.52e-	03 -2.20e-01
LCD incineration	Sulfur oxides	secondary	-1.75e-	02 -5.91e-01
Total End-of-life			-2.97e-	02 -1.00e+00
Total All Life-cycle Stages			2.96e+	00 1.00e+02
I able MI-21.	CKI LCIA Results IU	the All Falticulates In	ipaci Category	
---	---------------------	-------------------------	--------------------------	------------
Process Group	Materials	LCI Data Type	Air Particulates (kg)	% of total
Materials Processing Life-cyc	le Stage			
Steel Prod., cold-rolled, semi-finished	PM	secondary	1.05e-01	3.48e+01
Aluminum Prod.	PM	secondary	8.90e-03	2.96e+00
Polycarbonate Production	PM	secondary	6.46e-03	2.15e+00
Invar	PM	secondary	2.76e-03	9.15e-01
Ferrite mfg.	PM	secondary	1.72e-03	5.72e-01
ABS Production	PM	secondary	1.27e-03	4.22e-01
Styrene-butadiene Copolymer Prod.	PM	secondary	9.93e-04	3.30e-01
Lead	PM	secondary	8.41e-04	2.79e-01
Polystyrene Prod., high-impact	PM	secondary	3.02e-04	1.00e-01
Total Materials Processing			1.28e-01	4.25e+01
Manufacturing Life-cycle Sta	ge			
LPG Production	PM	secondary	1.29e-01	4.28e+01
Japanese Electric Grid	PM-10	model/secondary	2.00e-03	6.64e-01
Fuel Oil #6 Prod.	РМ	secondary	9.24e-04	3.07e-01
US electric grid	PM-10	model/secondary	9.22e-04	3.06e-01
Natural Gas Prod.	РМ	secondary	4.14e-04	1.38e-01
Fuel Oil #2 Prod.	PM	secondary	3.96e-04	1.31e-01
LPG Production	PM-10	secondary	2.25e-04	7.47e-02
Glass/frit	PM	primary	1.09e-04	3.63e-02
Fuel Oil #4 Prod.	PM	secondary	4.05e-05	1.34e-02
Fuel Oil #6 Prod.	PM-10	secondary	2.33e-06	7.74e-04
Fuel Oil #2 Prod.	PM-10	secondary	7.42e-07	2.47e-04
Glass/frit	PM	primary	6.67e-07	2.21e-04
Fuel Oil #4 Prod.	PM-10	secondary	8.70e-08	2.89e-05
Natural Gas Prod.	PM-10	secondary	4.94e-09	1.64e-06
Total Manufacturing			1.34e-01	4.45e+01
Use, Maintenance and Repair	· Life-cycle Stage			
US electric grid	PM-10	model/secondary	5.78e-02	1.92e+01
Total Use, Maintenance and F	Repair		5.78e-02	1.92e+01
End-of-life Life-cycle Stage				
CRT landfilling	PM	primary	2.08e-04	6.92e-02
US electric grid	PM-10	model/secondary	5.78e-06	1.92e-03
LPG Production	PM	secondary	1.11e-06	3.70e-04
LPG Production	PM-10	secondary	1.94e-09	6.45e-07
Natural Gas Prod.	PM-10	secondary	-2.48e-09	-8.23e-07
CRT Incineration	PM-10	secondary	-6.30e-08	-2.09e-05
Fuel Oil #4 Prod.	PM-10	secondary	-9.35e-07	-3.11e-04
Natural Gas Prod.	PM	secondary	-2.08e-04	-6.90e-02
Fuel Oil #4 Prod.	PM	secondary	-4.35e-04	-1.44e-01
CRT Incineration	PM	secondary	-1.83e-02	-6.09e+00
Total End-of-life			-1.88e-02	-6.23e+00
Total All Life-cycle Stages			3.01e-01	1.00e+02

Table M-21. CRT LCIA Results for the Air Particulates Impact Category

Process Group	Materials	LCI Data Type	Air Particulates (kg)	% of Total
Materials Processing Life-cycle Stage				
Steel Prod., cold-rolled, semi-finished	PM	secondary	5.14e-02	4.48e+01
Natural Gas Prod.	PM	secondary	2.89e-02	2.52e+01
Polycarbonate Production	PM	secondary	3.61e-03	3.15e+00
Aluminum Prod.	PM	secondary	3.31e-03	2.89e+00
PMMA Sheet Prod.	PM	secondary	3.22e-03	2.81e+00
PET Resin Production	PM	secondary	6.99e-04	6.10e-01
Styrene-butadiene Copolymer Prod.	PM	secondary	4.34e-04	3.79e-01
Natural Gas Prod.	PM-10	secondary	3.45e-07	3.01e-04
Total Materials Processing			9.16e-02	7.99e+01
Manufacturing Life-cycle Stage				
Japanese Electric Grid	PM-10	model/secondary	6.72e-03	5.87e+00
LPG Production	PM	secondary	6.19e-03	5.40e+00
Natural Gas Prod.	PM	secondary	6.53e-04	5.69e-01
US electric grid	PM-10	model/secondary	1.12e-04	9.76e-02
Fuel Oil #4 Prod.	PM	secondary	6.24e-05	5.44e-02
Fuel Oil #6 Prod.	PM	secondary	3.15e-05	2.75e-02
Panel components	PM	primary	2.74e-05	2.39e-02
Fuel Oil #2 Prod.	PM	secondary	1.85e-05	1.61e-02
Monitor/module	PM	primary	1.10e-05	9.58e-03
LPG Production	PM-10	secondary	1.08e-05	9.42e-03
LCD glass mfg.	PM	primary	5.06e-06	4.41e-03
Fuel Oil #4 Prod.	PM-10	secondary	1.34e-07	1.17e-04
Fuel Oil #6 Prod.	PM-10	secondary	7.94e-08	6.92e-05
Fuel Oil #2 Prod.	PM-10	secondary	3.47e-08	3.02e-05
Natural Gas Prod.	PM-10	secondary	7.79e-09	6.79e-06
Total Manufacturing			1.38e-02	1.21e+01
Use, Maintenance and Repair Life-cycle Stage				
US electric grid	PM-10	model/secondary	2.16e-02	1.88e+01
Total Use, Maintenance and Repair			2.16e-02	1.88e+01
End-of-life Life-cycle Stage				
LCD landfilling	PM	primary	5.02e-05	4.38e-02
US electric grid	PM-10	model/secondary	4.09e-06	3.57e-03
LPG Production	PM	secondary	5.08e-07	4.43e-04
Natural Gas Prod.	PM-10	secondary	-1.63e-09	-1.43e-06
LCD incineration	PM-10	secondary	-4.08e-08	-3.56e-05
Fuel Oil #4 Prod.	PM-10	secondary	-6.18e-07	-5.39e-04
Natural Gas Prod.	PM	secondary	-1.37e-04	-1.20e-01
Fuel Oil #4 Prod.	PM	secondary	-2.87e-04	-2.51e-01
LCD incineration	PM	secondary	-1.20e-02	-1.05e+01
Total End-of-life			-1.24e-02	-1.08e+01
Total All Life-cycle Stages			1.15e-01	1.00e+02

Table M-22. LCD LCIA Results for the Air P	Particulates Impact Category
--	------------------------------

Process Group	Material	LCI Data Type	Water Eutrophication (kg Phosphate Equivalents)	% of Total
Materials Processing Life-cycle Stag	e			
Steel Prod., cold-rolled, semi-finished	Other nitrogen	secondary	1.80e-04	3.73e-01
Steel Prod., cold-rolled, semi-finished	COD	secondary	1.61e-04	3.34e-01
Aluminum Prod.	Phosphate (PO43-)	secondary	9.44e-05	1.96e-01
Polycarbonate Production	Phosphate (PO43-)	secondary	4.84e-05	1.00e-01
ABS Production	Ammonia	secondary	3.70e-05	7.67e-02
Lead	Phosphate (PO43-)	secondary	2.79e-05	5.79e-02
ABS Production	COD	secondary	2.05e-05	4.25e-02
Polycarbonate Production	COD	secondary	2.03e-05	4.22e-02
ABS Production	Other nitrogen	secondary	1.78e-05	3.69e-02
Steel Prod., cold-rolled, semi-finished	Ammonia	secondary	1.68e-05	3.48e-02
Aluminum Prod.	Other nitrogen	secondary	1.23e-05	2.56e-02
Steel Prod., cold-rolled, semi-finished	Phosphorus (yellow or white)	secondary	1.20e-05	2.48e-02
Aluminum Prod.	Nitrogen	secondary	1.10e-05	2.28e-02
Invar	Other nitrogen	secondary	5.96e-06	1.24e-02
Invar	COD	secondary	5.80e-06	1.20e-02
Ferrite mfg.	COD	secondary	5.66e-06	1.18e-02
Invar	Ammonia	secondary	4.91e-06	1.02e-02
Aluminum Prod.	Nitrate	secondary	4.56e-06	9.47e-03
Styrene-butadiene Copolymer Prod.	COD	secondary	4.10e-06	8.50e-03
Lead	Nitrogen	secondary	3.77e-06	7.83e-03
Polycarbonate Production	Nitrogen	secondary	3.49e-06	7.24e-03
Lead	Other nitrogen	secondary	3.06e-06	6.36e-03
Ferrite mfg.	Phosphate (PO43-)	secondary	3.04e-06	6.31e-03
ABS Production	Nitrate	secondary	3.01e-06	6.24e-03
Styrene-butadiene Copolymer Prod.	Other nitrogen	secondary	1.74e-06	3.61e-03
Invar	Nitrates/nitrites	secondary	1.62e-06	3.37e-03
Ferrite mfg.	Ammonia	secondary	1.58e-06	3.29e-03
Steel Prod., cold-rolled, semi-finished	Phosphate (PO43-)	secondary	1.44e-06	2.98e-03
Ferrite mfg.	Other nitrogen	secondary	1.27e-06	2.65e-03
Polystyrene Prod., high-impact	COD	secondary	1.20e-06	2.48e-03
Aluminum Prod.	COD	secondary	1.18e-06	2.44e-03
Styrene-butadiene Copolymer Prod.	Ammonia ions	secondary	1.17e-06	2.42e-03
Lead	Nitrate	secondary	9.00e-07	1.87e-03
Polycarbonate Production	Ammonia	secondary	7.11e-07	1.48e-03
Steel Prod., cold-rolled, semi-finished	Nitrates/nitrites	secondary	6.71e-07	1.39e-03
Polystyrene Prod., high-impact	Nitrogen	secondary	5.08e-07	1.05e-03
Ferrite mfg.	Nitrate	secondary	4.04e-07	8.38e-04
Polystyrene Prod., high-impact	Ammonia	secondary	3.10e-07	6.44e-04
Lead	COD	secondary	2.92e-07	6.07e-04
Styrene-butadiene Copolymer Prod.	Nitrate	secondary	2.90e-07	6.01e-04
Ferrite mfg.	Phosphorus (yellow or white)	secondary	5.76e-08	1.20e-04
Invar	Phosphorus (yellow or white)	secondary	4.85e-08	1.01e-04
ABS Production	Phosphate (PO43-)	secondary	4.62e-08	9.59e-05
Polycarbonate Production	Nitrate	secondary	4.62e-08	9.58e-05
Polystyrene Prod., high-impact	Nitrate	secondary	3.02e-08	6.27e-05
Total Materials Processing			7.22e-04	1.50e+00
Manufacturing Life-cycle Stage				

Table M-23. CRT LCIA Results for the Water Eutrophication Impact Category

Process Group	Material	LCI Data Type	Water Eutrophication (kg Phosphate Equivalents)	% of Total
LPG Production	COD	secondary	3.49e-02	7.24e+01
LPG Production	Ammonia ions	secondary	9.04e-03	1.88e+01
CRT tube mfg.	Nitrogen	primary	3.02e-03	6.26e+00
CRT tube mfg.	COD	primary	1.59e-04	3.30e-01
CRT tube mfg.	Phosphorus (yellow or white)	primary	1.54e-04	3.21e-01
Fuel Oil #6 Prod.	COD	secondary	1.18e-04	2.45e-01
Fuel Oil #2 Prod.	COD	secondary	9.79e-05	2.03e-01
Fuel Oil #6 Prod.	Ammonia ions	secondary	3.01e-05	6.24e-02
Fuel Oil #2 Prod.	Ammonia ions	secondary	2.51e-05	5.21e-02
Fuel Oil #4 Prod.	COD	secondary	7.96e-06	1.65e-02
LPG Production	Nitrate	secondary	6.06e-06	1.26e-02
Fuel Oil #4 Prod.	Ammonia ions	secondary	2.04e-06	4.23e-03
CRT tube mfg.	Phosphate as P2O5	primary	1.62e-06	3.36e-03
Natural Gas Prod.	COD	secondary	7.15e-07	1.49e-03
Glass/frit	Nitrates/nitrites	primary	3.95e-07	8.21e-04
Glass/frit	COD	primary	1.81e-07	3.75e-04
Natural Gas Prod.	Ammonia ions	secondary	1.11e-07	2.30e-04
Fuel Oil #6 Prod.	Nitrate	secondary	2.43e-08	5.04e-05
Fuel Oil #2 Prod.	Nitrate	secondary	1.73e-08	3.58e-05
LPG Production	Other nitrogen	secondary	6.62e-09	1.37e-05
Natural Gas Prod.	Nitrate	secondary	1.50e-09	3.12e-06
Fuel Oil #4 Prod.	Nitrate	secondary	1.47e-09	3.05e-06
Fuel Oil #6 Prod.	Other nitrogen	secondary	2.24e-11	4.65e-08
Fuel Oil #2 Prod.	Other nitrogen	secondary	1.86e-11	3.85e-08
Fuel Oil #4 Prod.	Other nitrogen	secondary	1.51e-12	3.13e-09
Natural Gas Prod.	Other nitrogen	secondary	1.34e-13	2.78e-10
Total Manufacturing			4 760 02	0.970±01
Use Maintenance and Densiry Life a			4./00-02	9.0/0+01
Use, Maintenance and Repair Life-c	ycle Stage			
Total Use, Maintenance and Repair			0.00e+00	0.00e+00
End-of-life Life-cycle Stage				
CRT landfilling	COD	primary	2.13e-06	4.43e-03
CRT landfilling	Ammonia	primary	4.62e-07	9.59e-04
LPG Production	COD	secondary	3.01e-07	6.25e-04
LPG Production	Ammonia ions	secondary	7.80e-08	1.62e-04
LPG Production	Nitrate	secondary	5.23e-11	1.09e-07
LPG Production	Other nitrogen	secondary	5.71e-14	1.19e-10
Natural Gas Prod.	Other nitrogen	secondary	-6.71e-14	-1.39e-10
CRT Incineration	Other nitrogen	secondary	-1.04e-12	-2.16e-09
Fuel Oil #4 Prod.	Other nitrogen	secondary	-1.62e-11	-3.37e-08
Natural Gas Prod.	Nitrate	secondary	-7.54e-10	-1.57e-06
Fuel Oil #4 Prod.	Nitrate	secondary	-1.58e-08	-3.27e-05
Natural Gas Prod.	Ammonia ions	secondary	-5.56e-08	-1.15e-04
CRT Incineration	Nitrate	secondary	-1.80e-07	-3.73e-04
Natural Gas Prod.	COD	secondary	-3.59e-07	-7.45e-04
CRT Incineration	Ammonia	secondary	-1.80e-06	-3.74e-03
CRT Incineration	COD	secondary	-3.21e-06	-6.65e-03
Fuel Oil #4 Prod.	Ammonia ions	secondary	-2.19e-05	-4.55e-02

Table M-23. CRT LCIA Results for the Water Eutrophication Impact Category

Process Group	Material	LCI Data Type	Water Eutrophication (kg Phosphate Equivalents)	% of Total
Fuel Oil #4 Prod.	COD	secondary	-8.56e-05	-1.78e-01
Total End-of-life			-1.10e-04	-2.29e-01
Total All Manufacturing Stages			4.82e-02	1.00e+02

Table M-23. CRT LCIA Results for the Water Eutrophication Impact Category

Process Group	Material	LCI Data Type	Water Eutrophication (kg Phosphate Equivalents)	% of Total
Materials Processing Life-cycle Stag	e			
PMMA Sheet Prod.	Ammonia	secondary	1.97e-04	3.97e-01
Steel Prod., cold-rolled, semi-finished	Other nitrogen	secondary	8.81e-05	1.78e-01
Steel Prod., cold-rolled, semi-finished	COD	secondary	7.89e-05	1.59e-01
Natural Gas Prod.	COD	secondary	5.00e-05	1.01e-01
PMMA Sheet Prod.	Phosphate (PO43-)	secondary	4.86e-05	9.80e-02
Aluminum Prod.	Phosphate (PO43-)	secondary	3.51e-05	7.09e-02
Polycarbonate Production	Phosphate (PO43-)	secondary	2.70e-05	5.45e-02
PMMA Sheet Prod.	COD	secondary	2.19e-05	4.43e-02
Polycarbonate Production	COD	secondary	1.13e-05	2.29e-02
Steel Prod., cold-rolled, semi-finished	Ammonia	secondary	8.21e-06	1.66e-02
Natural Gas Prod.	Ammonia ions	secondary	7.75e-06	1.56e-02
PMMA Sheet Prod.	Nitrates/nitrites	secondary	7.67e-06	1.55e-02
Steel Prod., cold-rolled, semi-finished	Phosphorus (yellow or white)	secondary	5.86e-06	1.18e-02
Aluminum Prod.	Other nitrogen	secondary	4.60e-06	9.27e-03
PET Resin Production	COD	secondary	4.39e-06	8.86e-03
Aluminum Prod.	Nitrogen	secondary	4.09e-06	8.25e-03
Polycarbonate Production	Nitrogen	secondary	1.95e-06	3.93e-03
Styrene-butadiene Copolymer Prod.	COD	secondary	1.79e-06	3.61e-03
Aluminum Prod.	Nitrate	secondary	1.70e-06	3.43e-03
PMMA Sheet Prod.	Other nitrogen	secondary	9.66e-07	1.95e-03
Styrene-butadiene Copolymer Prod.	Other nitrogen	secondary	7.60e-07	1.53e-03
Steel Prod., cold-rolled, semi-finished	Phosphate (PO43-)	secondary	7.03e-07	1.42e-03
Styrene-butadiene Copolymer Prod.	Ammonia ions	secondary	5.11e-07	1.03e-03
Aluminum Prod.	COD	secondary	4.38e-07	8.83e-04
Polycarbonate Production	Ammonia	secondary	3.97e-07	8.01e-04
Steel Prod., cold-rolled, semi-finished	Nitrates/nitrites	secondary	3.29e-07	6.63e-04
Styrene-butadiene Copolymer Prod.	Nitrate	secondary	1.27e-07	2.55e-04
Natural Gas Prod.	Nitrate	secondary	1.05e-07	2.12e-04
PET Resin Production	Other nitrogen	secondary	3.81e-08	7.69e-05
Polycarbonate Production	Nitrate	secondary	2.58e-08	5.20e-05
PET Resin Production	Ammonia ions	secondary	2.33e-08	4.70e-05
PET Resin Production	Nitrate	secondary	9.08e-09	1.83e-05
Natural Gas Prod.	Other nitrogen	secondary	9.34e-12	1.88e-08
Total Materials Processing			6.10e-04	1.23e+00
Manufacturing Life-cycle Stage				11200 . 00
Monitor/module	Nitrogon	nrimory	2 220 02	6 722+01
Monitor/module	Phosphorus (vellow or white)	primary	1 320 02	0.72e+01
I PG Production	COD	secondary	1.520-02	2.000+01
LPG Production	Ammonia iong	secondary	1.076-03	8 75 - 01
Panel components	Nitrogen	primary	4.346-04 2.40e.04	4 84e 01
Panal components	Phosphorus (vallow or white)	primary	7.60° 02	1.530.01
Manitar/module	COD	primary	5.010.05	1.556-01
Panel components	COD	primary	1 860 05	0.81e.02
Fuel Oil #4 Prod	COD	secondary	4.800-05	2.47e 02
Fuel Oil #2 Prod	COD	secondary	1.230-03	0.222.02
Fuel Oil #6 Prod	COD	secondary	4.570-00	9.220-03 8.11e-03
Fuel Oil #4 Prod	Ammonia ions	secondary	3 1/2 06	6 330 02
Fuel Oil #2 Prod	Ammonia jons	secondary	1 170 06	2 360 02
$1 uvi 0 11 \pi 2 1100.$	i sinifionia ions	secondary	1.176-00	2.500-05

Table M-24. LCD LCIA Results for the Water Eutrophication Impact Category

Process Group	Material	LCI Data Type	Water Eutrophication (kg Phosphate Equivalents)	% of Total
Natural Gas Prod.	COD	secondary	1.13e-06	2.28e-03
Fuel Oil #6 Prod.	Ammonia ions	secondary	1.02e-06	2.07e-03
LPG Production	Nitrate	secondary	2.91e-07	5.86e-04
Natural Gas Prod.	Ammonia ions	secondary	1.75e-07	3.53e-04
LCD glass mfg.	Nitrate	primary	1.83e-08	3.70e-05
LCD glass mfg.	COD	primary	8.36e-09	1.69e-05
Natural Gas Prod.	Nitrate	secondary	2.37e-09	4.78e-06
Fuel Oil #4 Prod.	Nitrate	secondary	2.26e-09	4.56e-06
Fuel Oil #6 Prod.	Nitrate	secondary	8.26e-10	1.67e-06
Fuel Oil #2 Prod.	Nitrate	secondary	8.06e-10	1.63e-06
LPG Production	Other nitrogen	secondary	3.17e-10	6.41e-07
Fuel Oil #4 Prod.	Other nitrogen	secondary	2.33e-12	4.69e-09
Fuel Oil #2 Prod.	Other nitrogen	secondary	8.67e-13	1.75e-09
Fuel Oil #6 Prod.	Other nitrogen	secondary	7.62e-13	1.54e-09
Natural Gas Prod.	Other nitrogen	secondary	2.11e-13	4.25e-10
Total Manufacturing			4.90e-02	9.89e+01
Use, Maintenance and Repair Life-cy	ycle Stage			
Total Use, Maintenance and Repair			0.00e+00	0.00e+00
End-of-life Life-cycle Stage				
LCD landfilling	COD	primary	5.14e-07	1.04e-03
LPG Production	COD	secondary	1.37e-07	2.77e-04
LCD landfilling	Ammonia	primary	1.11e-07	2.25e-04
LPG Production	Ammonia ions	secondary	3.56e-08	7.19e-05
LPG Production	Nitrate	secondary	2.39e-11	4.81e-08
LPG Production	Other nitrogen	secondary	2.61e-14	5.26e-11
Natural Gas Prod.	Other nitrogen	secondary	-4.43e-14	-8.93e-11
LCD incineration	Other nitrogen	secondary	-6.75e-13	-1.36e-09
Fuel Oil #4 Prod.	Other nitrogen	secondary	-1.07e-11	-2.16e-08
Natural Gas Prod.	Nitrate	secondary	-4.98e-10	-1.00e-06
Fuel Oil #4 Prod.	Nitrate	secondary	-1.04e-08	-2.10e-05
Natural Gas Prod.	Ammonia ions	secondary	-3.67e-08	-7.41e-05
LCD incineration	Nitrate	secondary	-1.17e-07	-2.35e-04
Natural Gas Prod.	COD	secondary	-2.37e-07	-4.78e-04
LCD incineration	Ammonia	secondary	-1.40e-06	-2.83e-03
LCD incineration	COD	secondary	-3.16e-06	-6.38e-03
Fuel Oil #4 Prod.	Ammonia ions	secondary	-1.45e-05	-2.92e-02
Fuel Oil #4 Prod.	COD	secondary	-5.65e-05	-1.14e-01
Total End-of-life			-7.51e-05	-1.52e-01
Total All Life-cycle Stages			4.96e-02	1.00e+02

Table M-24. LCD LCIA Results for the Water Eutrophication Impact Category

Process Group	Material	LCI Data Type	BOD (kg) %	of Total
Materials Processing Life-cycle Stage				
Steel Prod., cold-rolled, semi-finished	BOD	secondary	2.51e-04	1.29e-01
Polycarbonate Production	BOD	secondary	8.68e-05	4.45e-02
Styrene-butadiene Copolymer Prod.	BOD	secondary	1.65e-05	8.48e-03
ABS Production	BOD	secondary	1.40e-05	7.17e-03
Polystyrene Prod., high-impact	BOD	secondary	6.80e-06	3.49e-03
Invar	BOD	secondary	6.65e-06	3.41e-03
Ferrite mfg.	BOD	secondary	4.74e-06	2.43e-03
Aluminum Prod.	BOD	secondary	4.03e-06	2.06e-03
Lead	BOD	secondary	2.48e-06	1.27e-03
Total Materials Processing			3.93e-04	2.02e-01
Manufacturing Life-cycle Stage				
LPG Production	BOD	secondary	1.88e-01	9.61e+01
CRT tube mfg.	BOD	primary	6.39e-03	3.28e+00
Fuel Oil #6 Prod.	BOD	secondary	6.34e-04	3.25e-01
Fuel Oil #2 Prod.	BOD	secondary	5.26e-04	2.70e-01
Fuel Oil #4 Prod.	BOD	secondary	4.28e-05	2.19e-02
Glass/frit	BOD	primary	8.20e-06	4.21e-03
Natural Gas Prod.	BOD	secondary	4.00e-06	2.05e-03
Total Manufacturing			1.95e-01	1.00e+02
Use, Maintenance and Repair Life-cycle Stage				
Total Use, Maintenance and Repair			0.00e+00	0.00e+00
End-of-life Life-cycle Stage				
CRT landfilling	BOD	primary	1.17e-05	5.98e-03
LPG Production	BOD	secondary	1.62e-06	8.30e-04
Natural Gas Prod.	BOD	secondary	-2.01e-06	-1.03e-03
CRT Incineration	BOD	secondary	-1.70e-05	-8.74e-03
Fuel Oil #4 Prod.	BOD	secondary	-4.60e-04	-2.36e-01
Total End-of-life			-4.65e-04	-2.39e-01
Total All Life-cycle Stages			1.95e-01	1.00e+02

Table M-25. CRT LCIA Results for the Water Quality, BOD Impact Category

	tor the water	Quanty, DOD II	ipace catego	<u> </u>
Process Group	Material	LCI Data Type	BOD (kg)	% of Total
Materials Processing Life-cycle Stage				
Natural Gas Prod.	BOD	secondary	2.80e-04	9.88e-01
PMMA Sheet Prod.	BOD	secondary	2.30e-04	8.12e-01
Steel Prod., cold-rolled, semi-finished	BOD	secondary	1.23e-04	4.34e-01
PET Resin Production	BOD	secondary	8.17e-05	2.88e-01
Polycarbonate Production	BOD	secondary	4.85e-05	1.71e-01
Styrene-butadiene Copolymer Prod.	BOD	secondary	7.23e-06	2.55e-02
Aluminum Prod.	BOD	secondary	1.50e-06	5.29e-03
Total Materials Processing			7.72e-04	2.72e+00
Manufacturing Life-cycle Stage				
Monitor/module	BOD	primary	1.74e-02	6.15e+01
LPG Production	BOD	secondary	9.00e-03	3.18e+01
Panel components	BOD	primary	1.34e-03	4.74e+00
Fuel Oil #4 Prod.	BOD	secondary	6.59e-05	2.33e-01
Fuel Oil #2 Prod.	BOD	secondary	2.46e-05	8.67e-02
Fuel Oil #6 Prod.	BOD	secondary	2.16e-05	7.62e-02
Natural Gas Prod.	BOD	secondary	6.31e-06	2.23e-02
LCD glass mfg.	BOD	primary	3.80e-07	1.34e-03
Total Manufacturing			2.79e-02	9.84e+01
Use, Maintenance and Repair Life-cycle Stage				
Total Use, Maintenance and Repair			0.00e+00	0.00e+00
End-of-life Life-cycle Stage				
LCD landfilling	BOD	primary	2.81e-06	9.91e-03
LPG Production	BOD	secondary	7.39e-07	2.61e-03
Natural Gas Prod.	BOD	secondary	-1.33e-06	-4.68e-03
LCD incineration	BOD	secondary	-1.70e-05	-5.99e-02
Fuel Oil #4 Prod.	BOD	secondary	-3.04e-04	-1.07e+00
Total End-of-life			-3.18e-04	-1.12e+00
Total All Life-cycle Stages			2.83e-02	1.00e+02

Table M-26. LCD LCIA Results for the Water Quality, BOD Impact Category

Process Group	Material	LCI Data Type	Total Suspended Solids (kg)	% of Total
Materials Processing Life-cycle Stage				
Steel Prod., cold-rolled, semi-finished	Suspended solids	secondary	2.04e-03	2.34e-01
Aluminum Prod.	Suspended solids	secondary	1.74e-03	1.99e-01
Polycarbonate Production	Suspended solids	secondary	1.11e-03	1.27e-01
ABS Production	Suspended solids	secondary	1.02e-03	1.16e-01
Invar	Suspended solids	secondary	8.30e-04	9.49e-02
Lead	Suspended solids	secondary	3.80e-04	4.34e-02
Styrene-butadiene Copolymer Prod.	Suspended solids	secondary	3.06e-04	3.50e-02
Ferrite mfg.	Suspended solids	secondary	2.45e-04	2.81e-02
Polystyrene Prod., high-impact	Suspended solids	secondary	5.14e-05	5.88e-03
Total Materials Processing			7.72e-03	8.83e-01
Manufacturing Life-cycle Stage				
LPG Production	Suspended solids	secondary	8.51e-01	9.74e+01
Glass/frit	Suspended solids	primary	7.23e-03	8.26e-01
CRT tube mfg.	Suspended solids	primary	4.63e-03	5.30e-01
Fuel Oil #6 Prod.	Suspended solids	secondary	2.88e-03	3.29e-01
Fuel Oil #2 Prod.	Suspended solids	secondary	2.39e-03	2.73e-01
Fuel Oil #4 Prod.	Suspended solids	secondary	1.94e-04	2.22e-02
Japanese Electric Grid	Suspended solids	model/secondary	2.27e-05	2.60e-03
Natural Gas Prod.	Suspended solids	secondary	1.72e-05	1.97e-03
Total Manufacturing			8.69e-01	9.94e+01
Use, Maintenance and Repair Life-cycle St	age			
Total Use, Maintenance and Repair			0.00e+00	0.00e+00
End-of-life Life-cycle Stage				
CRT landfilling	Suspended solids	primary	5.52e-05	6.31e-03
LPG Production	Suspended solids	secondary	7.35e-06	8.40e-04
Natural Gas Prod.	Suspended solids	secondary	-8.64e-06	-9.88e-04
CRT Incineration	Suspended solids	secondary	-7.55e-05	-8.63e-03
Fuel Oil #4 Prod.	Suspended solids	secondary	-2.09e-03	-2.39e-01
Total End-of-life			-2.11e-03	-2.41e-01
Total All Life-cycle Stages			8.74e-01	1.00e+02

Table M-27. CRT LCIA Results for the Water Quality, TSS Impact Category

Process Group	Material	LCI Data Type	Total Suspended Solids (kg) % of	Total
Materials Processing Life-cycle Stage				
PMMA Sheet Prod.	Suspended solids	secondary	1.34e-03	2.18e+00
Natural Gas Prod.	Suspended solids	secondary	1.20e-03	1.96e+00
Steel Prod., cold-rolled, semi-finished	Suspended solids	secondary	1.00e-03	1.63e+00
Aluminum Prod.	Suspended solids	secondary	6.48e-04	1.05e+00
Polycarbonate Production	Suspended solids	secondary	6.19e-04	1.01e+00
Styrene-butadiene Copolymer Prod.	Suspended solids	secondary	1.34e-04	2.18e-01
PET Resin Production	Suspended solids	secondary	3.45e-05	5.61e-02
Total Materials Processing			4.98e-03	8.10e+00
Manufacturing Life-cycle Stage				
LPG Production	Suspended solids	secondary	4.09e-02	6.64e+01
Monitor/module	Suspended solids	primary	1.55e-02	2.52e+01
Panel components	Suspended solids	primary	6.46e-04	1.05e+00
LCD glass mfg.	Suspended solids	primary	3.35e-04	5.44e-01
Fuel Oil #4 Prod.	Suspended solids	secondary	2.99e-04	4.86e-01
Fuel Oil #2 Prod.	Suspended solids	secondary	1.12e-04	1.81e-01
Fuel Oil #6 Prod.	Suspended solids	secondary	9.81e-05	1.59e-01
Japanese Electric Grid	Suspended solids	model/secondary	7.63e-05	1.24e-01
Natural Gas Prod.	Suspended solids	secondary	2.72e-05	4.42e-02
Total Manufacturing			5.80e-02	9.42e+01
Use, Maintenance and Repair Life-cycle	Stage			
Total Use, Maintenance and Repair			0.00e+00	0.00e+00
End-of-life Life-cycle Stage				
LCD landfilling	Suspended solids	primary	1.33e-05	2.16e-02
LPG Production	Suspended solids	secondary	3.35e-06	5.45e-03
Natural Gas Prod.	Suspended solids	secondary	-5.71e-06	-9.28e-03
LCD incineration	Suspended solids	secondary	-7.69e-05	-1.25e-01
Fuel Oil #4 Prod.	Suspended solids	secondary	-1.38e-03	-2.24e+00
Total End-of-life			-1.44e-03	-2.35e+00
Total All Life-cycle Stages			6.15e-02	1.00e+02

Naterials Processing Life-cycle Stage Steel Prod., cold-rolled, semi-finished Plutonium-241 (isotope) secondary 6.88e+66 1.77e+01 Farrite mag. Plutonium-241 (isotope) secondary 0.64e+06 1.77e+01 Steel Prod., cold-rolled, semi-finished Plutonium-241 (isotope) secondary 0.38e+04 2.36e-01 Steel Prod., cold-rolled, semi-finished Ratonium-230 (isotope) secondary 7.37e+04 7.32e-02 Steel Prod., cold-rolled, semi-finished Ratonium-230 (isotope) secondary 2.38e+04 7.44e-02 Invar Plutonium-240 (isotope) secondary 2.38e+04 7.44e-02 Invar C (sium-15) (isotope) secondary 2.38e+04 6.38e-02 Invar Radon-222 (isotope) secondary 2.48e+04 6.58e-02 Invar Radon-222 (isotope) secondary 2.48e+04 6.38e-02 Invar Radon-222 (isotope) secondary 2.48e+04 6.38e-02 Invar Radon-222 (isotope) secondary 3.8e+04 3.64e-02 Invar Radon-222 (isotope) <th>Process Group</th> <th>Material</th> <th>LCI Data Type</th> <th>Radioactivity (Bq) %</th> <th>of Total</th>	Process Group	Material	LCI Data Type	Radioactivity (Bq) %	of Total
Stack Prod., cold-rolled, semi-finished Pharainan-241 (isotope) secondary 6.86e+06 1.77e+01 Invar Pharainan-241 (isotope) secondary 6.86e+06 1.77e+01 Steck Prod., cold-rolled, semi-finished Pharainan-241 (isotope) secondary 9.38e+04 2.46e+01 Steck Prod., cold-rolled, semi-finished Pharainan-220 (isotope) secondary 2.97e+04 2.28e+01 Steck Prod., cold-rolled, semi-finished Pharainan-220 (isotope) secondary 2.98e+04 7.45e+02 Invar Pharainan-230 (isotope) secondary 2.86e+04 6.45e+02 Ferrite mfg. Caisum-135 (isotope) secondary 2.48e+04 6.34e+02 Invar Radon-222 (isotope) secondary 2.06e+04 6.34e+02 Ferrite mfg. Pharainan-299 (isotope) secondary 2.06e+04 6.34e+02 Invar Pharainan-290 (isotope) secondary 2.06e+04 6.34e+02 Invar Radon-222 (isotope) secondary 3.07e+03 1.03e+02 Steck Prod., cold-rolled, semi-finished Trotinan-90 (isotope) second	Materials Processing Life-cycle	Stage			
InvarPlucinim-241 (istopp)secondary6.680-061.72e-01Ferrite nfg.Plucinim-240 (istopp)secondary1.040-052.690-01Steel Prod., cold-rolled, semi-finishedPlutonim-230 (istopp)secondary8.77e-042.38e-01Steel Prod., cold-rolled, semi-finishedRubor.232 (istopp)secondary2.78e-047.38e-04Steel Prod., cold-rolled, semi-finishedRubor.232 (istopp)secondary2.98e-047.62-02Steel Prod., cold-rolled, semi-finishedRubor.232 (istopp)secondary2.87e-047.62-02Ferrite nfg.Plutonim.240 (istopp)secondary2.87e-046.78-02InvarPlutonim.220 (istopp)secondary2.48e-046.67s-02InvarRador.222 (istopp)secondary2.48e-046.53e-02InvarRador.222 (istopp)secondary2.06e-045.32e-02InvarPlutonim.230 (istopp)secondary2.06e-045.36e-02InvarPlutonim.230 (istopp)secondary3.87e-031.016-02InvarStorotim.90 (istopp)secondary3.87e-031.016-02Steel Prod., cold-rolled, semi-finishedTratimari (istopp)secondary3.87e-031.016-02Steel Prod., cold-rolled, semi-finishedradicarbey (seu synthege)secondary3.87e-021.88e-03Steel Prod., cold-rolled, semi-finishedradicarbey (seu synthege)secondary3.87e-021.88e-03Steel Prod., cold-rolled, semi-finishedradicarbey (seu synthege)secondary3.87e-02 <td>Steel Prod., cold-rolled, semi-finished</td> <td>Plutonium-241 (isotope)</td> <td>secondary</td> <td>2.40e+07</td> <td>6.23e+01</td>	Steel Prod., cold-rolled, semi-finished	Plutonium-241 (isotope)	secondary	2.40e+07	6.23e+01
Ferrite mfg. Pultonium-241 (stotope) secondary 6.64e+06 1.72e+01 Steel Prod., cold-rolled, semi-finished Cesium-135 (tostope) secondary 9.84e+04 2.43e-01 Steel Prod., cold-rolled, semi-finished Radon-222 (tostope) secondary 7.29e+04 2.43e-01 Steel Prod., cold-rolled, semi-finished Platonium-230 (tostope) secondary 2.38e+04 7.32e+02 Invar Platonium-240 (tostope) secondary 2.38e+04 7.32e+02 Invar Platonium-240 (tostope) secondary 2.38e+04 7.43e+02 Invar Radon-222 (tostope) secondary 2.48e+04 6.35e-02 Ferrite mfg. Radon-222 (tostope) secondary 2.48e+04 6.36e-02 Invar Platonium-329 (tostope) secondary 2.08e+04 6.36e-02 Invar Platonium-329 (tostope) secondary 3.37e+03 1.101e-02 Steel Prod., cold-rolled, semi-finished Tritium-30 (tostope) secondary 3.37e+03 1.016+02 Steel Prod., cold-rolled, semi-finished Tritium-30 (tostope) secondary <td>Invar</td> <td>Plutonium-241 (isotope)</td> <td>secondary</td> <td>6.80e+06</td> <td>1.77e+01</td>	Invar	Plutonium-241 (isotope)	secondary	6.80e+06	1.77e+01
Sted Prod., cold-rolled, semi-finished Planoium-240 (isotopp) secondary 9.38e+04 2.43e-01 Sted Prod., cold-rolled, semi-finished Rator-222 (isotopp) secondary 2.74e+04 2.138e-01 Ivar Platonium-240 (isotopp) secondary 2.94e+04 7.62-02 Ferrite angle, Platonium-240 (isotopp) secondary 2.94e+04 7.62-02 Ivar Cesium-135 (isotopp) secondary 2.48e+04 6.745-02 Ivar Radon-222 (isotopp) secondary 2.48e+04 6.736-02 Ivar Radon-222 (isotopp) secondary 1.016-02 Ivar Radon-222 (isotopp) secondary 1.016-02 Ivar Storotium-390 (isotopp) secondary 3.76e+03 <td>Ferrite mfg.</td> <td>Plutonium-241 (isotope)</td> <td>secondary</td> <td>6.64e+06</td> <td>1.72e+01</td>	Ferrite mfg.	Plutonium-241 (isotope)	secondary	6.64e+06	1.72e+01
Steel Prod., cold-rolled, semi-finished Radon-222 (isotope) secondary 9.38:04 2.23:e-01 Steel Prod., cold-rolled, semi-finished Radon-222 (isotope) secondary 7.28:e-04 1.89:e-01 Invar Platonium-240 (isotope) secondary 2.87:e-04 1.89:e-01 Invar Platonium-240 (isotope) secondary 2.87:e-04 6.89:e-02 Ferrite mfg. Cesium-153 (isotope) secondary 2.48:e-04 6.73:e-02 Ferrite mfg. Radon-222 (isotope) secondary 2.48:e-04 6.73:e-02 Ferrite mfg. Platonium-230 (isotope) secondary 2.06:e-04 6.36:e-02 Ferrite mfg. Platonium-230 (isotope) secondary 2.06:e-04 5.3:e-02 Extel Prod., cold-rolled, semi-finished Strontium-90 (isotope) secondary 3.97:e-03 1.03:e-02 Steel Prod., cold-rolled, semi-finished Trotium-90 (isotope) secondary 3.87:e-03 4.04:e-03 Steel Prod., cold-rolled, semi-finished Trotium-30 (isotope) secondary 3.87:e-03 4.04:e-03 Steel Prod., cold-rolled, semi-finished	Steel Prod., cold-rolled, semi-finished	Plutonium-240 (isotope)	secondary	1.04e+05	2.69e-01
Sted Prod, cold-rolled, semi-finished Radom-222 (istopp) secondary 7.29e-H0 1.89e-U1 Invar Plutonium-240 (istope) secondary 2.94r-H0 7.62e-U2 Ferrite mfg. Plutonium-240 (istope) secondary 2.87r-H0 7.46e-U2 Invar Cesium-135 (istopp) secondary 2.48r-H0 6.78-02 Invar Cesium-135 (istopp) secondary 2.48r-H0 6.73-02 Invar Radom-222 (istopp) secondary 2.48r-H0 6.73-02 Invar Plutonium-230 (istopc) secondary 2.46r-H0 5.36e-U2 Invar Plutonium-230 (istopc) secondary 3.76r-H0 1.04r-H0 Invar Strontium-90 (istopc) secondary 3.76r-H0 1.04r-H0 Invar Strontium-90 (istopc) secondary 3.78r-H0 1.04r-H0 Steel Prod, cold-rolled, semi-finished Tritium-3 (istopp) secondary 3.78r-H0 4.04e-H0 Steel Prod, cold-rolled, semi-finished Radioactive gas (unspecified) secondary 5.8r-H0 4.04e-H0 Ste	Steel Prod., cold-rolled, semi-finished	Cesium-135 (isotope)	secondary	9.38e+04	2.43e-01
Stecl Prod., cold-rolled, semi-finished Plutonium-239 (isotope) secondary 2,34:-04 7,82:-01 Invar Plutonium-240 (isotope) secondary 2,37:-04 7,416-02 Invar Cesium-135 (isotope) secondary 2,35:-04 6,3802 Invar Radon-222 (isotope) secondary 2,48:+04 6,45:-02 Invar Radon-222 (isotope) secondary 2,48:+04 6,45:-02 Invar Plutonium-239 (isotope) secondary 2,06:-04 5,36:-02 Steel Prod., cold-rolled, semi-finished Strontium-90 (isotope) secondary 3,37:+03 1,06:-02 Steel Prod., cold-rolled, semi-finished Tratioum-90 (isotope) secondary 3,37:+03 1,06:-02 Steel Prod., cold-rolled, semi-finished Tratioum-90 (isotope) secondary 1,38:+03 4,76:e03 Steel Prod., cold-rolled, semi-finished Tratioum-90 (isotope) secondary 1,38:+03 4,76:e03 Steel Prod., cold-rolled, semi-finished Tratious-13 (isotope) secondary 1,38:-03 4,04:e03 Steel Prod., cold-rolled, semi-finished	Steel Prod., cold-rolled, semi-finished	Radon-222 (isotope)	secondary	8.77e+04	2.28e-01
Invar Plutonium-240 (isotope) secondary 2.487-04 7.62-02 Ferrite mfg. Cesim-135 (isotope) secondary 2.687-04 6.809-02 Ferrite mfg. Cesim-135 (isotope) secondary 2.487-04 6.639-02 Invar Radon-222 (isotope) secondary 2.487-04 6.639-02 Ferrite mfg. Radon-222 (isotope) secondary 2.016-04 5.36e-02 Ferrite mfg. Plutonium-230 (isotope) secondary 1.040-04 5.64-02 Steel Prod., cold-rolled, semi-finished Strontium-90 (isotope) secondary 3.87c+03 1.010-02 Steel Prod., cold-rolled, semi-finished Tritium-3 (isotope) secondary 1.86-03 4.04e-04 Steel Prod., cold-rolled, semi-finished Radioactive substance (unspecified) secondary 1.86-03 4.04e-03 Steel Prod., cold-rolled, semi-finished Radioactive substance (unspecified) secondary 1.86-03 4.04e-03 Steel Prod., cold-rolled, semi-finished Radiom-226 (isotope) secondary 1.86-03 4.04e-02 Steel Prod., cold-rolled, semi-finished	Steel Prod., cold-rolled, semi-finished	Plutonium-239 (isotope)	secondary	7.29e+04	1.89e-01
Ferrite mfg. Platonium-240 (isotope) secondary 2.87c+04 7.44c-04 Invar Cesium-135 (isotope) secondary 2.65c+04 6.89c02 Invar Radon-222 (isotope) secondary 2.46c+04 6.43e-02 Invar Platonium-230 (isotope) secondary 2.46c+04 6.30e-02 Invar Platonium-230 (isotope) secondary 2.01e+04 5.23e-02 Steel Prod., cold-rolled, semi-finished Strontium-90 (isotope) secondary 3.97c+03 1.01e-02 Steel Prod., cold-rolled, semi-finished Tridium-30 (isotope) secondary 1.83e+03 4.76e-03 Steel Prod., cold-rolled, semi-finished Tridium-30 (isotope) secondary 1.83e+03 4.76e-03 Steel Prod., cold-rolled, semi-finished Tadioactive gas (unspecified) secondary 1.83e+03 4.76e-03 Steel Prod., cold-rolled, semi-finished Kanoni-33 (isotope) secondary 8.77e+02 2.28e-03 Invar Tridium-3 (isotope) secondary 5.9e+02 1.8e-03 Invar Tridium-3 (isotope) secondary	Invar	Plutonium-240 (isotope)	secondary	2.94e+04	7.62e-02
Invar Cesiun-135 (isotope) secondary 2.65er14 6.789e.02 Ferrite mfg. Cesiun-135 (isotope) secondary 2.48er144 6.736-02 Invar Radon-222 (isotope) secondary 2.48er144 6.30e-02 Invar Plutonium-239 (isotope) secondary 2.01er44 6.30e-02 Sted Prod., cold-rolled, seni-finished Strontium-90 (isotope) secondary 1.01e-02 Sted Prod., cold-rolled, seni-finished Strontium-90 (isotope) secondary 2.50e-03 6.50e-03 Sted Prod., cold-rolled, seni-finished Tritium-30 (isotope) secondary 1.82er03 4.01e-03 Sted Prod., cold-rolled, seni-finished Xenor-133 (isotope) secondary 8.76e-03 4.01e-03 Sted Prod., cold-rolled, seni-finished Radioactive substance (unspecified) secondary 8.76e-03 2.28e-03 Sted Prod., cold-rolled, seni-finished Radioactive substance (unspecified) secondary 5.96e-02 1.84e-03 Invar radioactive gas (unspecified) secondary 5.96e-02 1.84e-03 Invar radioactive gas (unspecif	Ferrite mfg.	Plutonium-240 (isotope)	secondary	2.87e+04	7.44e-02
Ferrite mfg. Cesium-155 (stope) secondary 2.59e+04 6.73e-02 Invar Radon-222 (isotope) secondary 2.44e+04 6.45e-02 Errite mfg. Radon-222 (isotope) secondary 2.06e+04 6.30e-02 Invar Plutonium-239 (isotope) secondary 2.06e+04 5.36e-02 Ferrite mfg. Plutonium-200 (isotope) secondary 3.97e+03 1.03e-02 Steel Prod., cold-rolled, semi-finished Strontium-90 (isotope) secondary 3.87e+03 1.01e-02 Steel Prod., cold-rolled, semi-finished Tritium-3 (isotope) secondary 1.83e+03 4.76e-03 Steel Prod., cold-rolled, semi-finished Tradioartive substance (unspecified) secondary 1.87e+03 4.76e-03 Steel Prod., cold-rolled, semi-finished Radioarcive substance (unspecified) secondary 8.76e+02 2.28e-03 Steel Prod., cold-rolled, semi-finished Theium-3 (isotope) secondary 5.09e+02 1.84e-03 Invar Tritium-3 (isotope) secondary 5.09e+02 1.84e-03 Invar radioactive gas (unspecified)<	Invar	Cesium-135 (isotope)	secondary	2.65e+04	6.89e-02
InvarRadon-222 (isotope)secondary2.48e+046.45e-02Ferrite mfg.Radon-222 (isotope)secondary2.40e+046.30e-02InvarPlutonim-239 (isotope)secondary2.06e+045.36e-02Steel Prod., cold-rolled, semi-finishedStrontium-90 (isotope)secondary1.40e+043.64e-02InvarStrontium-90 (isotope)secondary3.78r+031.10e+02Ferrite mfg.Strontium-90 (isotope)secondary3.78r+031.10e+02Steel Prod., cold-rolled, semi-finishedTritium-3 (isotope)secondary1.83e+034.76e+03Steel Prod., cold-rolled, semi-finishedXenon-133 (isotope)secondary1.85e+034.47ee+03Steel Prod., cold-rolled, semi-finishedRadioactive gas (unspecified)secondary8.76e+022.28e+03Steel Prod., cold-rolled, semi-finishedRadiaut-226 (isotope)secondary7.09e+021.84e+03InvarTritium-3 (isotope)secondary5.19e+021.35e+03InvarTritium-3 (isotope)secondary5.19e+021.35e+03Invarradioactive gas (unspecified)secondary5.19e+021.32e+03InvarXenon-133 (isotope)secondary4.30e+021.12e+03Steel Prod., cold-rolled, semi-finishedThorium-230 (isotope)secondary4.30e+021.12e+03InvarXenon-133 (isotope)secondary4.30e+021.12e+03Steel Prod., cold-rolled, semi-finishedThorium-230 (isotope)secondary4.30e+021.12e+03<	Ferrite mfg.	Cesium-135 (isotope)	secondary	2.59e+04	6.73e-02
Ferrite mfg. Radon-222 (isotope) secondary 2.42e+04 6.30e-02 Invar Plutonium-239 (isotope) secondary 2.06e+04 5.36e-02 Steel Prod., cold-rolled, semi-finished Strontium-90 (isotope) secondary 3.97e+03 1.00e+04 Ferrite mfg. Strontium-90 (isotope) secondary 3.97e+03 1.00e+02 Steel Prod., cold-rolled, semi-finished Tritium-3 (isotope) secondary 2.50e+03 6.50e+03 Steel Prod., cold-rolled, semi-finished Tritium-3 (isotope) secondary 1.83e+03 4.76e+03 Steel Prod., cold-rolled, semi-finished radioactive substance (unspecified) secondary 8.77e+02 2.28e+03 Steel Prod., cold-rolled, semi-finished Radioactive substance (unspecified) secondary 8.76e+02 2.58e+03 Steel Prod., cold-rolled, semi-finished Radioactive substance (unspecified) secondary 7.09e+02 2.85e+03 Steel Prod., cold-rolled, semi-finished Trotium-230 (isotope) secondary 5.19e+02 1.35e+03 Steel Prod., cold-rolled, semi-finished Trotium-230 (isotope) secondary 3.19e+02	Invar	Radon-222 (isotope)	secondary	2.48e+04	6.45e-02
InvarPlutonium-239 (isotope)secondary2.06e+045.36e-02Ferrite mfg.Plutonium-239 (isotope)secondary2.01e+045.23e-02Ited Prod., cold-rolled, semi-finishedStrontium-90 (isotope)secondary3.97e+031.03e-02InvarStrontium-90 (isotope)secondary3.87e+031.01e-02Steel Prod., cold-rolled, semi-finishedTritium-30 (isotope)secondary2.56e+036.50e+03Steel Prod., cold-rolled, semi-finishedTritium-3 (isotope)secondary1.83e+034.76e-03Steel Prod., cold-rolled, semi-finishedXanon-133 (isotope)secondary1.85e+034.47ee-03Steel Prod., cold-rolled, semi-finishedXanon-133 (isotope)secondary8.76e+022.25e+03InvarTritium-3 (isotope)secondary7.09e+021.84e-03InvarTritium-3 (isotope)secondary5.19e+021.84e-03InvarTritium-3 (isotope)secondary5.19e+021.34e-03InvarXanon-133 (isotope)secondary5.19e+021.34e-03InvarXanon-133 (isotope)secondary5.07e+021.32e-03InvarXanon-133 (isotope)secondary3.91e+021.12e-03Steel Prod., cold-rolled, semi-finishedThorium-244 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedUraium-244 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedUraium-244 (isotope)secondary1.92e+041.11e+0	Ferrite mfg.	Radon-222 (isotope)	secondary	2.42e+04	6.30e-02
Ferrite mfg.Plutonium-239 (isotope)secondary2.01e+045.23e-02Steel Prod., cold-rolled, semi-finishedStrontium-90 (isotope)secondary1.40e+043.64e-02Ferrite mfg.Strontium-90 (isotope)secondary3.87e+031.01e-02Steel Prod., cold-rolled, semi-finishedTritium-3 (isotope)secondary2.30e+036.50e-03Steel Prod., cold-rolled, semi-finishedTritium-3 (isotope)secondary1.83e+034.76e-03Steel Prod., cold-rolled, semi-finishedXenon-133 (isotope)secondary8.77e+022.28e-03InvarRadioactive gas (unspecified)secondary8.66e+022.28e-03InvarTritium-3 (isotope)secondary7.09e+021.84e-03Ferrite mfg.Tritium-3 (isotope)secondary5.19e+021.35e-03InvarTritium-3 (isotope)secondary5.19e+021.33e-03Steel Prod., cold-rolled, semi-finishedThorium-230 (isotope)secondary5.19e+021.32e-03Invarradioactive gas (unspecified)secondary5.07e+021.32e-03Steel Prod., cold-rolled, semi-finishedThorium-240 (isotope)secondary3.01e+021.12e-03Steel Prod., cold-rolled, semi-finishedPlutonium-242 (isotope)secondary3.01e+021.12e-03Steel Prod., cold-rolled, semi-finishedPlutonium-242 (isotope)secondary3.01e+021.12e-03Steel Prod., cold-rolled, semi-finishedVaniou-244 (isotope)secondary3.01e+021.02e-03Steel Prod.,	Invar	Plutonium-239 (isotope)	secondary	2.06e+04	5.36e-02
Steel Prod., cold-rolled, semi-finishedStrontium-90 (isotope)secondary1.40e+043.64e-02InvarStrontium-90 (isotope)secondary3.87e+031.03e-02Steel Prod., cold-rolled, semi-finishedTritium-3 (isotope)secondary2.50e+036.50e-03Steel Prod., cold-rolled, semi-finishedTritium-3 (isotope)secondary1.83e+034.76e-03Steel Prod., cold-rolled, semi-finishedXenon-133 (isotope)secondary8.77e+022.28e-03Steel Prod., cold-rolled, semi-finishedRadioactive gas (unspecified)secondary8.77e+022.28e-03InvarTritium-3 (isotope)secondary6.92e+021.84e-03InvarTritium-3 (isotope)secondary6.92e+021.84e-03Invarradioactive gas (unspecified)secondary5.19e+021.35e-03Steel Prod., cold-rolled, semi-finishedThorium-230 (isotope)secondary5.19e+021.35e-03Steel Prod., cold-rolled, semi-finishedThorium-230 (isotope)secondary5.19e+021.35e-03Steel Prod., cold-rolled, semi-finishedThorium-234 (isotope)secondary4.30e+021.12e-03InvarXenon-133 (isotope)secondary3.90e+021.10e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary3.90e+021.10e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary3.90e+021.10e-03Steel Prod., cold-rolled, semi-finishedAmericum-244 (isotope)secondary1.12e+02<	Ferrite mfg.	Plutonium-239 (isotope)	secondary	2.01e+04	5.23e-02
InvarStrontium-90 (isotope)secondary3.97e+031.03e-02Ferrite mfg.Strontium-90 (isotope)secondary3.87e+031.01e-02Steel Prod., cold-rolled, semi-finishedTritium-3 (isotope)secondary1.83e+034.76e-03Steel Prod., cold-rolled, semi-finishedRadioactive gas (unspecified)secondary1.83e+034.76e-03Steel Prod., cold-rolled, semi-finishedRadioactive gas (unspecified)secondary8.76e+022.28e-03InvarRadioactive gas (unspecified)secondary8.66e+022.25e-03InvarTritium-3 (isotope)secondary6.92e+021.84e-03Ferrite mfg.Tritium-3 (isotope)secondary5.19e+021.84e-03Invarradioactive gas (unspecified)secondary5.19e+021.35e-03Invarradioactive gas (unspecified)secondary5.19e+021.32e-03Invarradioactive gas (unspecified)secondary4.30e+021.12e-03InvarXenon-133 (isotope)secondary4.30e+021.12e-03Steel Prod., cold-rolled, semi-finishedPutonium-242 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary3.92e+026.21e-04Steel Prod., cold-rolled, semi-finishedCurium-244	Steel Prod., cold-rolled, semi-finished	Strontium-90 (isotope)	secondary	1.40e+04	3.64e-02
Ferrite mfg.Strontium-90 (isotope)secondary3.87e+031.01e-02Steel Prod., cold-rolled, semi-finishedTritium-3 (isotope)secondary2.50e+036.50e-03Steel Prod., cold-rolled, semi-finishedradioactive gas (unspecified)secondary1.85e+034.76e-03Steel Prod., cold-rolled, semi-finishedRadioactive substance (unspecified)secondary8.77e+022.28e-03Steel Prod., cold-rolled, semi-finishedRadioactive gas (unspecified)secondary8.66e+022.25e-03InvarTritium-3 (isotope)secondary6.92e+021.88e-03InvarTritium-3 (isotope)secondary5.19e+021.34e-03Invarradioactive gas (unspecified)secondary5.19e+021.34e-03Steel Prod., cold-rolled, semi-finishedThorium-230 (isotope)secondary5.07e+021.32e-03Invarradioactive gas (unspecified)secondary5.07e+021.32e-03InvarXanon-133 (isotope)secondary4.41e+021.12e-03Steel Prod., cold-rolled, semi-finishedPlutonium-242 (isotope)secondary3.90e+021.02e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary3.90e+021.02e-03Steel Prod., cold-rolled, semi-finishedAmericum-241 (isotope)secondary3.90e+021.02e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary1.92e+028.70e-04Steel Prod., cold-rolled, semi-finishedAmericum-241 (isotope)secondary1.2e+	Invar	Strontium-90 (isotope)	secondary	3.97e+03	1.03e-02
Steel Prod., cold-rolled, semi-finishedTritium-3 (isotope)secondary2.50e+036.50e-03Steel Prod., cold-rolled, semi-finishedXenon-133 (isotope)secondary1.83e+034.76e-03Steel Prod., cold-rolled, semi-finishedXenon-133 (isotope)secondary8.77e+022.22e-03InvarTritium-3 (isotope)secondary8.76e+022.25e-03InvarTritium-3 (isotope)secondary6.92e+021.84e-03InvarTritium-3 (isotope)secondary6.92e+021.80e-03Invarradioactive gas (unspecified)secondary5.19e+021.35e-03Steel Prod., cold-rolled, semi-finishedThorium-230 (isotope)secondary5.19e+021.32e-03Invarradioactive gas (unspecified)secondary5.07e+021.32e-03InvarXenon-133 (isotope)secondary4.41e+021.15e-03Steel Prod., cold-rolled, semi-finishedPutonium-242 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary3.92e+026.21e-04Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Radium-226 (isotope)secondary1.12e-033.15e+022.89e-04InvarPutonium-244 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Radium-226 (isotope)seco	Ferrite mfg.	Strontium-90 (isotope)	secondary	3.87e+03	1.01e-02
Steel Prod., cold-rolled, semi-finishedradioactive gas (unspecified)secondary1.83+034.76e-03Steel Prod., cold-rolled, semi-finishedXenon-133 (isotope)secondary8.77e+022.28e-03Steel Prod., cold-rolled, semi-finishedRadium-226 (isotope)secondary8.77e+1022.28e-03InvarTritium-3 (isotope)secondary6.92e+021.84e-03InvarTritium-3 (isotope)secondary6.92e+021.84e-03Invarradioactive gas (unspecified)secondary6.92e+021.84e-03Invarradioactive gas (unspecified)secondary5.19e+021.35e-03Steel Prod., cold-rolled, semi-finishedThorium-230 (isotope)secondary5.07e+021.32e-03Invarxenon-133 (isotope)secondary4.41e+021.15e-03Steel Prod., cold-rolled, semi-finishedPlutonium-242 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedUranium-244 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedUranium-244 (isotope)secondary3.15e+1028.19e-04Ferrite mfg.Radium-226 (isotope)secondary1.92e+024.90e-04Ferrite mfg.Thorium-230 (isotope)secondary1.11e+022.88e-04InvarCurium-244 (isotope)secondary1.11e+022.88e-04InvarCurium-244 (isotope)secondary1.11e+022.88e-04InvarCurium-244 (isotope)secondary1.10e+022.87e	Steel Prod., cold-rolled, semi-finished	Tritium-3 (isotope)	secondary	2.50e+03	6.50e-03
Steel Prod., cold-rolled, semi-finishedXenon-133 (isotope)secondary1.56e+034.04e-03Ferrite mfg.Radioactive substance (unspecified)secondary8.77e+022.28e-03Steel Prod., cold-rolled, semi-finishedRadium-226 (isotope)secondary8.66e+022.25e-03InvarTritium-3 (isotope)secondary6.92e+021.84e-03Ferrite mfg.Tritium-3 (isotope)secondary5.19e+021.35e-03Steel Prod., cold-rolled, semi-finishedThorium-230 (isotope)secondary5.19e+021.32e-03Invarradioactive gas (unspecified)secondary5.07e+021.32e-03Invarradioactive gas (unspecified)secondary4.10e+021.12e-03InvarXenon-133 (isotope)secondary4.30e+021.12e-03InvarXenon-133 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedUranim-244 (isotope)secondary3.90e+021.01e-03Steel Prod., cold-rolled, semi-finishedUranim-234 (isotope)secondary2.39e+026.1e-04Ferrite mfg.Radium-226 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Thorium-230 (isotope)secondary1.92e+023.90e+02Steel Prod., cold-rolled, semi-finishedKrapton-85 (isotope)secondary1.92e+023.90e+02Ferrite mfg.Thorium-230 (isotope)secondary1.92e+023.90e+022.80e-04InvarCurium-244 (isotope)secondary1.11e+022.	Steel Prod., cold-rolled, semi-finished	radioactive gas (unspecified)	secondary	1.83e+03	4.76e-03
Ferrite mfg.Radioactive substance (unspecified)secondary8.77e+022.28e-03Steel Prod., cold-rolled, semi-finishedRadium-226 (isotope)secondary7.09e+021.84e-03InvarTritium-3 (isotope)secondary6.92e+021.80e-03Invarradioactive gas (unspecified)secondary5.19e+021.35e-03Steel Prod., cold-rolled, semi-finishedThorium-230 (isotope)secondary5.19e+021.32e-03Invarradioactive gas (unspecified)secondary5.07e+021.32e-03InvarXenon-133 (isotope)secondary4.41e+021.15e-03InvarXenon-133 (isotope)secondary4.30e+021.12e-03InvarXenon-133 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedPlutonium-242 (isotope)secondary3.90e+021.11e-03Steel Prod., cold-rolled, semi-finishedUranium-234 (isotope)secondary2.39e+026.21e-04Steel Prod., cold-rolled, semi-finishedAmericium-241 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Radium-226 (isotope)secondary1.92e+023.70e-04Steel Prod., cold-rolled, semi-finishedAmericium-241 (isotope)secondary1.11e+022.88e-04InvarPlutonium-242 (isotope)secondary1.11e+022.88e-04InvarCurium-244 (isotope)secondary1.08e+022.71e-04Ferrite mfg.Curium-244 (isotope)secondary1.08e+022.81e-04 <t< td=""><td>Steel Prod., cold-rolled, semi-finished</td><td>Xenon-133 (isotope)</td><td>secondary</td><td>1.56e+03</td><td>4.04e-03</td></t<>	Steel Prod., cold-rolled, semi-finished	Xenon-133 (isotope)	secondary	1.56e+03	4.04e-03
Steel Prod. cold-rolled, semi-finishedRadium-226 (isotope)secondary8.66e+022.25e-03InvarTritium-3 (isotope)secondary7.09e+021.84e-03Ferrite mfg.Tritium-3 (isotope)secondary6.92e+021.80e-03Invarradioactive gas (unspecified)secondary5.19e+021.35e-03Steel Prod., cold-rolled, semi-finishedThorium-230 (isotope)secondary5.19e+021.32e-03Invarxationactive gas (unspecified)secondary5.07e+021.32e-03InvarXenon-133 (isotope)secondary4.30e+021.12e-03Steel Prod., cold-rolled, semi-finishedPlutonium-242 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary3.90e+021.01e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary2.39e+026.21e-04Steel Prod., cold-rolled, semi-finishedAmericium-226 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Radium-226 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Thorium-234 (isotope)secondary1.12e+033.70e-04InvarPlutonium-242 (isotope)secondary1.12e+023.70e-04InvarPlutonium-242 (isotope)secondary1.12e+023.70e-04InvarPlutonium-242 (isotope)secondary1.12e+023.70e-04InvarPlutonium-242 (isotope)secondary1.12e+022.8e-04Inv	Ferrite mfg.	Radioactive substance (unspecified)	secondary	8.77e+02	2.28e-03
InvarTritium-3 (isotope)secondary7.09e+021.84e-03Ferrite mfg.Tritium-3 (isotope)secondary6.92e+021.80e-03Invarradioactive gas (unspecified)secondary5.19e+021.35e-03Steel Prod., cold-rolled, semi-finishedThorium-230 (isotope)secondary5.15e+021.32e-03InvarXenon-133 (isotope)secondary5.07e+021.32e-03InvarXenon-133 (isotope)secondary4.30e+021.12e-03Ferrite mfg.Xenon-133 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary3.90e+021.02e-03Steel Prod., cold-rolled, semi-finishedUranium-234 (isotope)secondary3.15e+028.19e-04Ferrite mfg.Radium-226 (isotope)secondary1.92e+024.20e-03Steel Prod., cold-rolled, semi-finishedUranium-234 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Radium-226 (isotope)secondary1.12e+023.70e-04Steel Prod., cold-rolled, semi-finishedKrypton-85 (isotope)secondary1.12e+022.89e-04InvarPlutonium-244 (isotope)secondary1.12e+022.89e-04InvarPlutonium-244 (isotope)secondary1.11e+022.88e-04InvarPlutonium-244 (isotope)secondary1.08e+022.88e-04InvarCurium-244 (isotope)secondary1.08e+022.88e-04InvarRadium-226 (isotope)secon	Steel Prod., cold-rolled, semi-finished	Radium-226 (isotope)	secondary	8.66e+02	2.25e-03
Ferrite mfg.Tritium-3 (isotope)secondary6.92e+021.80e-03Invarradioactive gas (unspecified)secondary5.19e+021.35e-03Steel Prod., cold-rolled, semi-finishedThorium-230 (isotope)secondary5.15e+021.34e-03Ferrite mfg.radioactive gas (unspecified)secondary5.07e+021.32e-03InvarXenon-133 (isotope)secondary4.41e+021.12e-03Ferrite mfg.Xenon-133 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedPlutonium-244 (isotope)secondary3.90e+021.01e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary3.90e+021.01e-03Steel Prod., cold-rolled, semi-finishedRadium-226 (isotope)secondary3.90e+021.01e-03Steel Prod., cold-rolled, semi-finishedAmericium-241 (isotope)secondary1.92e+024.30e-04Ferrite mfg.Radium-226 (isotope)secondary1.92e+024.30e-04Ferrite mfg.Thorium-230 (isotope)secondary1.11e+022.88e-04InvarPlutonium-242 (isotope)secondary1.11e+022.88e-04InvarCurium-244 (isotope)secondary1.00e+022.87e-04Ferrite mfg.Plutonium-242 (isotope)secondary1.00e+022.87e-04InvarCurium-244 (isotope)secondary1.00e+022.87e-04InvarRadium-226 (isotope)secondary1.00e+022.87e-04InvarRadium-	Invar	Tritium-3 (isotope)	secondary	7.09e+02	1.84e-03
Invarradioactive gas (unspecified)secondary5.19e+021.35e-03Steel Prod., cold-rolled, semi-finishedThorium-230 (isotope)secondary5.15e+021.34e-03Ferrite mfg.radioactive gas (unspecified)secondary5.07e+021.32e-03InvarXenon-133 (isotope)secondary4.41e+021.12e-03Steel Prod., cold-rolled, semi-finishedPlutonium-242 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary3.90e+021.01e-03Steel Prod., cold-rolled, semi-finishedUranium-236 (isotope)secondary3.90e+021.01e-03Steel Prod., cold-rolled, semi-finishedUranium-246 (isotope)secondary3.90e+028.19e-04Ferrite mfg.Radium-226 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Thorium-230 (isotope)secondary1.92e+023.70e-04Steel Prod., cold-rolled, semi-finishedKrypton-85 (isotope)secondary1.11e+022.88e-04InvarPlutonium-242 (isotope)secondary1.11e+022.88e-04InvarCurium-244 (isotope)secondary1.00e+022.87e-04InvarCurium-242 (isotope)secondary1.08e+022.80e-04InvarPlutonium-242 (isotope)secondary1.08e+022.80e-04InvarRadium-226 (isotope)secondary1.08e+022.87e-04InvarRadium-226 (isotope)secondary1.02e+022.71e-04Steel P	Ferrite mfg.	Tritium-3 (isotope)	secondary	6.92e+02	1.80e-03
Steel Prod., cold-rolled, semi-finishedThorium-230 (isotope)secondary5.15e+021.34e-03Ferrite mfg.radioactive gas (unspecified)secondary5.07e+021.32e-03InvarXenon-133 (isotope)secondary4.41e+021.15e-03Ferrite mfg.Xenon-133 (isotope)secondary4.30e+021.12e-03Steel Prod., cold-rolled, semi-finishedPlutonium-242 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedUranium-234 (isotope)secondary3.90e+021.01e-03Steel Prod., cold-rolled, semi-finishedUranium-234 (isotope)secondary2.39e+026.21e-04Ferrite mfg.Radium-226 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Thorium-230 (isotope)secondary1.12e+023.70e-04Steel Prod., cold-rolled, semi-finishedAmericium-241 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Thorium-230 (isotope)secondary1.11e+022.88e-04InvarPlutonium-242 (isotope)secondary1.11e+022.88e-04InvarCurium-244 (isotope)secondary1.08e+022.81e-04Ferrite mfg.Plutonium-242 (isotope)secondary1.08e+022.81e-04InvarCurium-244 (isotope)secondary1.08e+022.80e-04InvarRadium-226 (isotope)secondary1.08e+022.80e-04InvarRadium-226 (isotope)secondary1.08e+022.80e-04InvarRadium-226 (Invar	radioactive gas (unspecified)	secondary	5.19e+02	1.35e-03
Ferrite mfg.radioactive gas (unspecified)secondary5.07e+021.32e-03InvarXenon-133 (isotope)secondary4.41e+021.15e-03Ferrite mfg.Xenon-133 (isotope)secondary4.30e+021.12e-03Steel Prod., cold-rolled, semi-finishedPlutonium-242 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary3.90e+021.01e-03Steel Prod., cold-rolled, semi-finishedUranium-234 (isotope)secondary2.39e+026.21e-04Steel Prod., cold-rolled, semi-finishedAmericium-241 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Radium-226 (isotope)secondary1.12e+023.70e-04Steel Prod., cold-rolled, semi-finishedAmericium-241 (isotope)secondary1.42e+023.70e-04Ferrite mfg.Thorium-230 (isotope)secondary1.11e+022.89e-04InvarPlutonium-242 (isotope)secondary1.10e+022.87e-04InvarCurium-244 (isotope)secondary1.08e+022.80e-04InvarCurium-244 (isotope)secondary1.08e+022.80e-04InvarRadium-226 (isotope)secondary1.02e+022.71e-04Steel Prod., cold-rolled, semi-finishedUranium-234 (isotope)secondary1.08e+022.80e-04InvarRadium-226 (isotope)secondary1.08e+022.87e-042.71e-04Ferrite mfg.Plutonium-242 (isotope)secondary1.02e+022.71e-0	Steel Prod., cold-rolled, semi-finished	Thorium-230 (isotope)	secondary	5.15e+02	1.34e-03
InvarXenon-133 (isotope)secondary4.41e+021.15e-03Ferrite mfg.Xenon-133 (isotope)secondary4.30e+021.12e-03Steel Prod., cold-rolled, semi-finishedPlutonium-242 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary3.90e+021.01e-03Steel Prod., cold-rolled, semi-finishedUranium-234 (isotope)secondary2.39e+026.21e-04Ferrite mfg.Radium-226 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Thorium-230 (isotope)secondary1.42e+023.70e-04Steel Prod., cold-rolled, semi-finishedKrypton-85 (isotope)secondary1.11e+022.88e-04InvarPlutonium-242 (isotope)secondary1.11e+022.88e-04InvarPlutonium-242 (isotope)secondary1.08e+022.81e-04InvarCurium-244 (isotope)secondary1.08e+022.81e-04InvarPlutonium-242 (isotope)secondary1.08e+022.81e-04InvarCurium-244 (isotope)secondary1.08e+022.81e-04InvarRadium-226 (isotope)secondary1.02e+022.80e-04InvarRadium-226 (isotope)secondary1.02e+022.80e-04InvarRadium-226 (isotope)secondary1.02e+022.80e-04InvarRadium-226 (isotope)secondary9.28e+012.41e-04InvarRadium-226 (isotope)secondary9.28e+012.21e-04 </td <td>Ferrite mfg.</td> <td>radioactive gas (unspecified)</td> <td>secondary</td> <td>5.07e+02</td> <td>1.32e-03</td>	Ferrite mfg.	radioactive gas (unspecified)	secondary	5.07e+02	1.32e-03
Ferrite mfg.Xenon-133 (isotope)secondary4.30e+021.12e-03Steel Prod., cold-rolled, semi-finishedPlutonium-242 (isotope)secondary3.91e+021.02e-03Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary3.90e+021.01e-03Steel Prod., cold-rolled, semi-finishedUranium-234 (isotope)secondary3.90e+021.01e-03Steel Prod., cold-rolled, semi-finishedUranium-234 (isotope)secondary2.39e+026.21e-04Ferrite mfg.Radium-226 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Thorium-230 (isotope)secondary1.12e+023.70e-04Steel Prod., cold-rolled, semi-finishedKrypton-85 (isotope)secondary1.11e+022.88e-04InvarPlutonium-242 (isotope)secondary1.11e+022.88e-04InvarPlutonium-242 (isotope)secondary1.08e+022.87e-04Ferrite mfg.Plutonium-242 (isotope)secondary1.08e+022.88e-04InvarRadium-226 (isotope)secondary1.08e+022.88e-04InvarRadium-226 (isotope)secondary1.08e+022.88e-04InvarRadium-226 (isotope)secondary1.08e+022.88e-04InvarRadium-226 (isotope)secondary1.08e+022.88e-04InvarRadium-226 (isotope)secondary1.08e+022.88e-04InvarRadium-226 (isotope)secondary1.08e+022.88e-04InvarRadium-226 (isotope)seconda	Invar	Xenon-133 (isotope)	secondary	4.41e+02	1.15e-03
Steel Prod., cold-rolled, semi-finishedPlutonium-242 (isotope)secondary $3.91e+02$ $1.02e-03$ Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary $3.90e+02$ $1.01e-03$ Steel Prod., cold-rolled, semi-finishedUranium-234 (isotope)secondary $3.15e+02$ $8.19e-04$ Ferrite mfg.Radium-226 (isotope)secondary $2.39e+02$ $6.21e-04$ Steel Prod., cold-rolled, semi-finishedAmericium-241 (isotope)secondary $1.92e+02$ $4.99e-04$ Ferrite mfg.Thorium-230 (isotope)secondary $1.42e+02$ $3.70e-04$ Steel Prod., cold-rolled, semi-finishedKrypton-85 (isotope)secondary $1.11e+02$ $2.89e-04$ InvarPlutonium-242 (isotope)secondary $1.11e+02$ $2.88e-04$ InvarCurium-244 (isotope)secondary $1.08e+02$ $2.81e-04$ Ferrite mfg.Plutonium-242 (isotope)secondary $1.08e+02$ $2.81e-04$ InvarRadium-226 (isotope)secondary $1.08e+02$ $2.81e-04$ InvarRadium-226 (isotope)secondary $1.08e+02$ $2.81e-04$ InvarRadium-226 (isotope)secondary $1.08e+02$ $2.81e-04$ InvarRadium-226 (isotope)secondary $1.02e+02$ $2.82e-04$ InvarRadium-226 (isotope)secondary $2.82e+01$ $2.41e-04$ InvarRadium-226 (isotope)secondary $2.82e+01$ $2.41e-04$ InvarUranium-234 (isotope)secondary $8.72e+01$ $2.26e-04$	Ferrite mfg.	Xenon-133 (isotope)	secondary	4.30e+02	1.12e-03
Steel Prod., cold-rolled, semi-finishedCurium-244 (isotope)secondary3.90e+021.01e-03Steel Prod., cold-rolled, semi-finishedUranium-234 (isotope)secondary3.15e+028.19e-04Ferrite mfg.Radium-226 (isotope)secondary2.39e+026.21e-04Steel Prod., cold-rolled, semi-finishedAmericium-241 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Thorium-230 (isotope)secondary1.42e+023.70e-04Steel Prod., cold-rolled, semi-finishedKrypton-85 (isotope)secondary1.11e+022.89e-04InvarPlutonium-242 (isotope)secondary1.10e+022.87e-04InvarCurium-244 (isotope)secondary1.08e+022.81e-04Ferrite mfg.Plutonium-242 (isotope)secondary1.08e+022.80e-04InvarCurium-244 (isotope)secondary1.08e+022.80e-04InvarRadium-226 (isotope)secondary1.04e+022.71e-04Steel Prod., cold-rolled, semi-finishedUranium-238 (isotope)secondary9.28e+012.41e-04InvarUranium-234 (isotope)secondary8.72e+012.26e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary8.67e+012.25e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary8.67e+012.25e-04InvarAmericium-241 (isotope)secondary8.67e+012.25e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondar	Steel Prod., cold-rolled, semi-finished	Plutonium-242 (isotope)	secondary	3.91e+02	1.02e-03
Steel Prod., cold-rolled, semi-finishedUranium-234 (isotope)secondary3.15e+028.19e-04Ferrite mfg.Radium-226 (isotope)secondary2.39e+026.21e-04Steel Prod., cold-rolled, semi-finishedAmericium-241 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Thorium-230 (isotope)secondary1.42e+023.70e-04Steel Prod., cold-rolled, semi-finishedKrypton-85 (isotope)secondary1.11e+022.89e-04InvarPlutonium-242 (isotope)secondary1.11e+022.88e-04InvarCurium-244 (isotope)secondary1.00e+022.87e-04Ferrite mfg.Plutonium-242 (isotope)secondary1.08e+022.81e-04Ferrite mfg.Curium-244 (isotope)secondary1.08e+022.80e-04InvarRadium-226 (isotope)secondary1.04e+022.71e-04Steel Prod., cold-rolled, semi-finishedUranium-238 (isotope)secondary9.28e+012.41e-04InvarUranium-234 (isotope)secondary8.72e+012.27e-04Steel Prod., cold-rolled, semi-finishedUranium-234 (isotope)secondary8.72e+012.27e-04Ferrite mfg.Uranium-234 (isotope)secondary8.72e+012.25e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary8.67e+012.25e-04Steel Prod., cold-rolled, semi-finishedNeptunium-237 (isotope)secondary5.31e+011.38e-04InvarAmericium-241 (isotope)secondary5.31e+01<	Steel Prod., cold-rolled, semi-finished	Curium-244 (isotope)	secondary	3.90e+02	1.01e-03
Ferrite mfg.Radium-226 (isotope)secondary2.39e+026.21e-04Steel Prod., cold-rolled, semi-finishedAmericium-241 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Thorium-230 (isotope)secondary1.42e+023.70e-04Steel Prod., cold-rolled, semi-finishedKrypton-85 (isotope)secondary1.11e+022.89e-04InvarPlutonium-242 (isotope)secondary1.11e+022.88e-04InvarCurium-244 (isotope)secondary1.10e+022.87e-04Ferrite mfg.Plutonium-242 (isotope)secondary1.08e+022.81e-04InvarCurium-244 (isotope)secondary1.08e+022.81e-04InvarRadium-226 (isotope)secondary1.04e+022.71e-04Steel Prod., cold-rolled, semi-finishedUranium-238 (isotope)secondary9.28e+012.41e-04InvarUranium-234 (isotope)secondary8.72e+012.26e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary8.67e+012.25e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary6.02e+011.56e-04InvarAmericium-237 (isotope)secondary5.31e+011.38e-04InvarAmericium-241 (isotope)secondary5.31e+011.38e-04InvarAmericium-241 (isotope)secondary5.31e+011.38e-04InvarAmericium-241 (isotope)secondary5.31e+011.38e-04InvarAmericium-241 (isotope)<	Steel Prod., cold-rolled, semi-finished	Uranium-234 (isotope)	secondary	3.15e+02	8.19e-04
Steel Prod., cold-rolled, semi-finishedAmericium-241 (isotope)secondary1.92e+024.99e-04Ferrite mfg.Thorium-230 (isotope)secondary1.42e+023.70e-04Steel Prod., cold-rolled, semi-finishedKrypton-85 (isotope)secondary1.11e+022.89e-04InvarPlutonium-242 (isotope)secondary1.11e+022.88e-04InvarCurium-244 (isotope)secondary1.10e+022.87e-04Ferrite mfg.Plutonium-242 (isotope)secondary1.08e+022.81e-04Ferrite mfg.Curium-244 (isotope)secondary1.08e+022.80e-04InvarRadium-226 (isotope)secondary1.04e+022.71e-04Steel Prod., cold-rolled, semi-finishedUranium-238 (isotope)secondary9.28e+012.41e-04InvarUranium-234 (isotope)secondary8.72e+012.26e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary8.67e+012.25e-04Steel Prod., cold-rolled, semi-finishedNeptunium-237 (isotope)secondary6.02e+011.56e-04InvarAmericium-241 (isotope)secondary5.44e+011.41e-04Ferrite mfg.Americium-241 (isotope)secondary5.31e+011.38e-04InvarKrypton-85 (isotope)secondary5.31e+011.38e-04	Ferrite mfg.	Radium-226 (isotope)	secondary	2.39e+02	6.21e-04
Ferrite mfg.Thorium-230 (isotope)secondary1.42e+023.70e-04Steel Prod., cold-rolled, semi-finishedKrypton-85 (isotope)secondary1.11e+022.89e-04InvarPlutonium-242 (isotope)secondary1.11e+022.88e-04InvarCurium-244 (isotope)secondary1.10e+022.87e-04Ferrite mfg.Plutonium-242 (isotope)secondary1.08e+022.81e-04Ferrite mfg.Curium-244 (isotope)secondary1.08e+022.80e-04InvarRadium-226 (isotope)secondary1.04e+022.71e-04Steel Prod., cold-rolled, semi-finishedUranium-238 (isotope)secondary9.28e+012.41e-04InvarUranium-234 (isotope)secondary8.74e+012.27e-04Ferrite mfg.Uranium-234 (isotope)secondary8.72e+012.26e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary8.67e+012.25e-04Steel Prod., cold-rolled, semi-finishedNeptunium-237 (isotope)secondary6.02e+011.56e-04InvarAmericium-241 (isotope)secondary5.44e+011.41e-04Ferrite mfg.Americium-241 (isotope)secondary5.31e+011.38e-04InvarKrypton-85 (isotope)secondary5.31e+011.38e-04	Steel Prod., cold-rolled, semi-finished	Americium-241 (isotope)	secondary	1.92e+02	4.99e-04
Steel Prod., cold-rolled, semi-finishedKrypton-85 (isotope)secondary1.11e+022.89e-04InvarPlutonium-242 (isotope)secondary1.11e+022.88e-04InvarCurium-244 (isotope)secondary1.10e+022.87e-04Ferrite mfg.Plutonium-242 (isotope)secondary1.08e+022.81e-04Ferrite mfg.Curium-244 (isotope)secondary1.08e+022.80e-04InvarRadium-226 (isotope)secondary1.04e+022.71e-04Steel Prod., cold-rolled, semi-finishedUranium-238 (isotope)secondary9.28e+012.41e-04InvarUranium-234 (isotope)secondary8.74e+012.27e-04Ferrite mfg.Uranium-234 (isotope)secondary8.72e+012.26e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary8.67e+012.25e-04Steel Prod., cold-rolled, semi-finishedNeptunium-237 (isotope)secondary6.02e+011.56e-04InvarAmericium-241 (isotope)secondary5.31e+011.38e-04InvarAmericium-241 (isotope)secondary5.31e+011.38e-04InvarAmericium-241 (isotope)secondary5.31e+011.38e-04InvarKrypton-85 (isotope)secondary5.31e+011.38e-04	Ferrite mfg.	Thorium-230 (isotope)	secondary	1.42e+02	3.70e-04
InvarPlutonium-242 (isotope)secondary1.11e+022.88e-04InvarCurium-244 (isotope)secondary1.10e+022.87e-04Ferrite mfg.Plutonium-242 (isotope)secondary1.08e+022.81e-04Ferrite mfg.Curium-244 (isotope)secondary1.08e+022.80e-04InvarRadium-226 (isotope)secondary1.04e+022.71e-04Steel Prod., cold-rolled, semi-finishedUranium-238 (isotope)secondary9.28e+012.41e-04InvarUranium-234 (isotope)secondary8.72e+012.26e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary8.67e+012.25e-04Steel Prod., cold-rolled, semi-finishedNeptunium-237 (isotope)secondary6.02e+011.56e-04InvarAmericium-241 (isotope)secondary5.44e+011.41e-04Ferrite mfg.Americium-241 (isotope)secondary5.31e+011.38e-04InvarKrypton-85 (isotope)secondary5.31e+018.18e-05	Steel Prod., cold-rolled, semi-finished	Krypton-85 (isotope)	secondary	1.11e+02	2.89e-04
InvarCurium-244 (isotope)secondary1.10e+022.87e-04Ferrite mfg.Plutonium-242 (isotope)secondary1.08e+022.81e-04Ferrite mfg.Curium-244 (isotope)secondary1.08e+022.80e-04InvarRadium-226 (isotope)secondary1.04e+022.71e-04Steel Prod., cold-rolled, semi-finishedUranium-238 (isotope)secondary9.28e+012.41e-04InvarUranium-234 (isotope)secondary8.74e+012.27e-04Ferrite mfg.Uranium-234 (isotope)secondary8.72e+012.26e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary8.67e+012.25e-04Steel Prod., cold-rolled, semi-finishedNeptunium-237 (isotope)secondary6.02e+011.56e-04InvarAmericium-241 (isotope)secondary5.44e+011.41e-04Ferrite mfg.Americium-241 (isotope)secondary5.31e+011.38e-04InvarKrypton-85 (isotope)secondary3.15e+018.18e-05	Invar	Plutonium-242 (isotope)	secondary	1.11e+02	2.88e-04
Ferrite mfg.Plutonium-242 (isotope)secondary1.08e+022.81e-04Ferrite mfg.Curium-244 (isotope)secondary1.08e+022.80e-04InvarRadium-226 (isotope)secondary1.04e+022.71e-04Steel Prod., cold-rolled, semi-finishedUranium-238 (isotope)secondary9.28e+012.41e-04InvarUranium-234 (isotope)secondary8.74e+012.27e-04Ferrite mfg.Uranium-234 (isotope)secondary8.72e+012.26e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary8.67e+012.25e-04Steel Prod., cold-rolled, semi-finishedNeptunium-237 (isotope)secondary6.02e+011.56e-04InvarAmericium-241 (isotope)secondary5.44e+011.41e-04Ferrite mfg.Americium-241 (isotope)secondary5.31e+011.38e-04InvarKrypton-85 (isotope)secondary3.15e+018.18e-05	Invar	Curium-244 (isotope)	secondary	1.10e+02	2.87e-04
Ferrite mfg.Curium-244 (isotope)secondary1.08e+022.80e-04InvarRadium-226 (isotope)secondary1.04e+022.71e-04Steel Prod., cold-rolled, semi-finishedUranium-238 (isotope)secondary9.28e+012.41e-04InvarUranium-234 (isotope)secondary8.74e+012.27e-04Ferrite mfg.Uranium-234 (isotope)secondary8.72e+012.26e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary8.67e+012.25e-04Steel Prod., cold-rolled, semi-finishedNeptunium-237 (isotope)secondary6.02e+011.56e-04InvarAmericium-241 (isotope)secondary5.44e+011.41e-04Ferrite mfg.Americium-241 (isotope)secondary5.31e+011.38e-04InvarKrypton-85 (isotope)secondary3.15e+018.18e-05	Ferrite mfg.	Plutonium-242 (isotope)	secondary	1.08e+02	2.81e-04
InvarRadium-226 (isotope)secondary1.04e+022.71e-04Steel Prod., cold-rolled, semi-finishedUranium-238 (isotope)secondary9.28e+012.41e-04InvarUranium-234 (isotope)secondary8.74e+012.27e-04Ferrite mfg.Uranium-234 (isotope)secondary8.72e+012.26e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary8.67e+012.25e-04Steel Prod., cold-rolled, semi-finishedNeptunium-237 (isotope)secondary6.02e+011.56e-04InvarAmericium-241 (isotope)secondary5.44e+011.41e-04Ferrite mfg.Americium-241 (isotope)secondary5.31e+011.38e-04InvarKrypton-85 (isotope)secondary3.15e+018.18e-05	Ferrite mfg.	Curium-244 (isotope)	secondary	1.08e+02	2.80e-04
Steel Prod., cold-rolled, semi-finishedUranium-238 (isotope)secondary9.28e+012.41e-04InvarUranium-234 (isotope)secondary8.74e+012.27e-04Ferrite mfg.Uranium-234 (isotope)secondary8.72e+012.26e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary8.67e+012.25e-04Steel Prod., cold-rolled, semi-finishedNeptunium-237 (isotope)secondary6.02e+011.56e-04InvarAmericium-241 (isotope)secondary5.44e+011.41e-04Ferrite mfg.Americium-241 (isotope)secondary5.31e+011.38e-04InvarKrypton-85 (isotope)secondary3.15e+018.18e-05	Invar	Radium-226 (isotope)	secondary	1.04e+02	2.71e-04
InvarUranium-234 (isotope)secondary8.74e+012.27e-04Ferrite mfg.Uranium-234 (isotope)secondary8.72e+012.26e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary8.67e+012.25e-04Steel Prod., cold-rolled, semi-finishedNeptunium-237 (isotope)secondary6.02e+011.56e-04InvarAmericium-241 (isotope)secondary5.44e+011.41e-04Ferrite mfg.Americium-241 (isotope)secondary5.31e+011.38e-04InvarKrypton-85 (isotope)secondary3.15e+018.18e-05	Steel Prod., cold-rolled, semi-finished	Uranium-238 (isotope)	secondary	9.28e+01	2.41e-04
Ferrite mfg.Uranium-234 (isotope)secondary8.72e+012.26e-04Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary8.67e+012.25e-04Steel Prod., cold-rolled, semi-finishedNeptunium-237 (isotope)secondary6.02e+011.56e-04InvarAmericium-241 (isotope)secondary5.44e+011.41e-04Ferrite mfg.Americium-241 (isotope)secondary5.31e+011.38e-04InvarKrypton-85 (isotope)secondary3.15e+018.18e-05	Invar	Uranium-234 (isotope)	secondary	8.74e+01	2.27e-04
Steel Prod., cold-rolled, semi-finishedSamarium-151 (isotope)secondary8.67e+012.25e-04Steel Prod., cold-rolled, semi-finishedNeptunium-237 (isotope)secondary6.02e+011.56e-04InvarAmericium-241 (isotope)secondary5.44e+011.41e-04Ferrite mfg.Americium-241 (isotope)secondary5.31e+011.38e-04InvarKrypton-85 (isotope)secondary3.15e+018.18e-05	Ferrite mfg.	Uranium-234 (isotope)	secondary	8.72e+01	2.26e-04
Steel Prod., cold-rolled, semi-finishedNeptunium-237 (isotope)secondary6.02e+011.56e-04InvarAmericium-241 (isotope)secondary5.44e+011.41e-04Ferrite mfg.Americium-241 (isotope)secondary5.31e+011.38e-04InvarKrypton-85 (isotope)secondary3.15e+018.18e-05	Steel Prod., cold-rolled, semi-finished	Samarium-151 (isotope)	secondary	8.67e+01	2.25e-04
InvarAmericium-241 (isotope)secondary5.44e+011.41e-04Ferrite mfg.Americium-241 (isotope)secondary5.31e+011.38e-04InvarKrypton-85 (isotope)secondary3.15e+018.18e-05	Steel Prod., cold-rolled, semi-finished	Neptunium-237 (isotope)	secondary	6.02e+01	1.56e-04
Ferrite mfg.Americium-241 (isotope)secondary5.31e+011.38e-04InvarKrypton-85 (isotope)secondary3.15e+018.18e-05	Invar	Americium-241 (isotope)	secondary	5.44e+01	1.41e-04
Invar Krypton-85 (isotope) secondary 3.15e+01 8.18e-05	Ferrite mfg.	Americium-241 (isotope)	secondary	5.31e+01	1.38e-04
	Invar	Krypton-85 (isotope)	secondary	3.15e+01	8.18e-05

 Table M-29. CRT LCIA Results for the Radioactivity Impact Category

Process Group	Material	LCI Data Type	Radioactivity (Ba) %	of Total
Ferrite mfg	Krypton-85 (isotone)	secondary	3 07e+01	7 99e-05
Ferrite mfg	Uranium-238 (isotope)	secondary	2 56e+01	6 66e-05
Invar	Samarium-151 (isotope)	secondary	2.45e+01	6 37e-05
Invar	Uranium-238 (isotope)	secondary	2.45e+01	6.35e-05
Ferrite mfg	Samarium-151 (isotope)	secondary	2.40e+01	6 22e-05
Steel Prod cold-rolled semi-finished	Carbon-14 (isotope)	secondary	1.91e+01	4 95e-05
Invar	Nentunium-237 (isotope)	secondary	1.71e+01	4.43e-05
Ferrite mfg	Nentunium-237 (isotope)	secondary	1.66e+01	4 32e-05
Invar	Radioactive substance (unspecified)	secondary	8 13e+00	2 11e-05
Steel Prod cold-rolled semi-finished	Radium-222 (isotone)	secondary	6.66e+00	1.73e-05
Steel Prod. cold-rolled semi-finished	Uranium-222 (isotope)	secondary	5.85e+00	1.75C-05
Invar	Carbon-14 (isotope)	secondary	5.40e+00	1.320 05
Ferrite mfg	Carbon-14 (isotope)	secondary	5.27e+00	1.40C-05
Invar	Thorium-230 (isotope)	secondary	5.10e+00	1.376-05
Steel Prod cold-rolled semi-finished	Americium-243 (isotope)	secondary	4 19e+00	1.09e-05
Steel Prod., cold-rolled, semi-finished	Tachnetium 99M (isotope)	secondary	2.86e+00	7.430.06
Steel Prod., cold-rolled, semi-finished	Therium 228 (isotope)	secondary	2.800+00	6 480 06
Steel Prod., cold-rolled, semi-finished	Paden 220 (isotope)	secondary	2.490+00	5 260 06
Inver	Radium 222 (isotope)	secondary	2.03e+00	1.050.06
Forrito mfg	Radium 222 (isotope)	secondary	1.910+00	4.956-00
Ferrite mig.	Lizanium 224 (isotope)	secondary	1.800+00	4.030-00
Invar	Uranium 228 (isotope)	secondary	1.800+00	4.630-00
Invar Forrito mfg	Uranium 225 (isotope)	secondary	1.620+00	4.070-00
Ferrite mig.	Uranium 225 (isotope)	secondary	1.620+00	4.200-00
Invar	Dedium 228 (isotope)	secondary	1.380+00	4.100-00
Steel Prod., cold-rolled, semi-linished	Americian 242 (isotope)	secondary	1.30e+00	2.0806
Invar	Americium 243 (isotope)	secondary	1.19e+00	3.08e-06
	Americium-245 (isotope)	secondary	1.100+00	3.000-00
Steel Prod., cold-rolled, semi-finished	Polonium-210 (isotope)	secondary	1.13e+00	2.92e-06
Invar	Technetium-99M (isotope)	secondary	8.10e-01	2.10e-06
Ferrite mig.	The improvement of the second	secondary	7.916-01	2.05e-06
Invar	Therium 228 (isotope)	secondary	7.52e-01	1.95e-06
		secondary	7.546-01	1.916-06
Steel Prod., cold-rolled, semi-finished	Lead-210 (isotope)	secondary	6.54e-01	1.70e-06
Steel Prod., cold-rolled, semi-finished	Radium-224 (isotope)	secondary	6.05e-01	1.57e-06
Invar	Radon-220 (isotope)	secondary	5.800-01	1.516-06
Ferrite mig.	Radon-220 (isotope)	secondary	5.666-01	1.4/e-06
Invar	Radium-228 (isotope)	secondary	3.90e-01	1.01e-06
Ferrite mfg.	Radium-228 (isotope)	secondary	3.81e-01	9.89e-07
Steel Prod., cold-rolled, semi-finished	Zirconium-93 (isotope)	secondary	3.76e-01	9.7/e-07
Invar	Polonium-210 (isotope)	secondary	3.21e-01	8.35e-07
Steel Prod., cold-rolled, semi-finished	Cesium-137 (isotope)	secondary	3.19e-01	8.27e-07
Ferrite mfg.	Polonium-210 (isotope)	secondary	3.14e-01	8.15e-07
Steel Prod., cold-rolled, semi-finished	Protactinium-234 (isotope)	secondary	2.01e-01	5.23e-07
Steel Prod., cold-rolled, semi-finished	Thorium-234 (isotope)	secondary	2.01e-01	5.23e-07
Steel Prod., cold-rolled, semi-finished	Silver-110M (isotope)	secondary	1.88e-01	4.88e-07
Invar	Lead-210 (isotope)	secondary	1.87e-01	4.85e-07
Invar	Radium-224 (isotope)	secondary	1.83e-01	4.75e-07
Ferrite mfg.	Lead-210 (isotope)	secondary	1.82e-01	4.73e-07
Ferrite mfg.	Radium-224 (isotope)	secondary	1.78e-01	4.63e-07
Steel Prod., cold-rolled, semi-finished	Potassium-40 (isotope)	secondary	1.75e-01	4.54e-07
Invar	Radium-226 (isotope)	secondary	1.41e-01	3.66e-07

Process Group Material LCI Data Type Radioactivity (Bq) %	of Total
Invar Thorium-230 (isotope) secondary 1.41e-01	3.66e-07
Steel Prod., cold-rolled, semi-finished Cobalt-58 (isotope) secondary 1.26e-01	3.27e-07
Steel Prod., cold-rolled, semi-finished Tin-126 (isotope) secondary 1.18e-01	3.07e-07
Invar Zirconium-93 (isotope) secondary 1.07e-01	2.77e-07
Ferrite mfg. Zirconium-93 (isotope) secondary 1.04e-01	2.70e-07
Ferrite mfg. Cesium-137 (isotope) secondary 8.80e-02	2.29e-07
Steel Prod., cold-rolled, semi-finished Cobalt-60 (isotope) secondary 7.90e-02	2.05e-07
Invar Uranium-235 (isotope) secondary 7.81e-02	2.03e-07
Invar Cesium-137 (isotope) secondary 7.42e-02	1.93e-07
Steel Prod., cold-rolled, semi-finished Selenium-79 (isotope) secondary 6.76e-02	1.75e-07
Radioactive aerosols and halogenes	1 49e-07
Invar Protectinium-234 (isotope) secondary 5.74-02	1.490-07
Invar Thorium 234 (isotope) secondary 5.70e-02	1.480.07
Exercise mfor the protectinium 224 (isotope) secondary 5.70-02	1.450.07
Former mig. Frotactimum-2.34 (isotope) secondary 5.57e-02	1.450.07
Ferrite inig. Inordun-254 (isotope) secondary 5.57e-02	1.456-07
Invar Silver-110M (isotope) secondary 5.52e-02	1.386-07
Ferrite mig. Silver-110M (isotope) secondary 5.19e-02	1.356-07
$\frac{1}{1000}$	1.30e-07
Ferrite mig. Potassium-40 (isotope) secondary 4.87e-02	1.2/e-0/
Steel Prod., cold-rolled, semi-finished Thorium-232 (isotope) secondary 4.65e-02	1.21e-07
Steel Prod., cold-rolled, semi-finished Curium-245 (isotope) secondary 4.35e-02	1.13e-07
Steel Prod., cold-rolled, semi-finished Antimony-124 (isotope) secondary 4.33e-02	1.12e-07
Steel Prod., cold-rolled, semi-finished Cesium-134 (isotope) secondary 3.88e-02	1.01e-07
Invar Cobalt-58 (isotope) secondary 3.5/e-02	9.26e-08
Ferrite mfg. Cobalt-58 (isotope) secondary 3.48e-02	9.04e-08
Invar Tin-126 (isotope) secondary 3.35e-02	8.69e-08
Ferrite mfg. Tin-126 (isotope) secondary 3.27e-02	8.48e-08
Invar Cobalt-60 (isotope) secondary 2.24e-02	5.81e-08
Ferrite mfg. Cobalt-60 (isotope) secondary 2.18e-02	5.67e-08
Steel Prod., cold-rolled, semi-finished Palladium-107 (isotope) secondary 2.11e-02	5.49e-08
Invar Selenium-79 (isotope) secondary 1.91e-02	4.97e-08
Ferrite mfg. Selenium-79 (isotope) secondary 1.87e-02	4.85e-08
Radioactive aerosols and halogenes	$4.22e_{-}08$
Invar (unspecificat) secondary 1.050-02	4 16e-08
Padioactive percents and halogenes	4.100-08
Ferrite mfg. (unspecified) secondary 1.59e-02	4.12e-08
Invar Thorium-232 (isotope) secondary 1.33e-02	3.45e-08
Ferrite mfg. Thorium-232 (isotope) secondary 1.30e-02	3.37e-08
Invar Curium-245 (isotope) secondary 1.23e-02	3.20e-08
Invar Antimony-124 (isotope) secondary 1.23e-02	3.18e-08
Ferrite mfg. Curium-245 (isotope) secondary 1.20e-02	3.12e-08
Ferrite mfg. Antimony-124 (isotope) secondary 1.20e-02	3.11e-08
Invar Cesium-134 (isotope) secondary 1 10e-02	2.85e-08
Ferrite mfg Cesium-134 (isotope) secondary 107e-02	2.79e-08
Steel Prod., cold-rolled, semi-finished Iodine-131 (isotope) secondary 9 03e-03	2.34e-08
Steel Prod., cold-rolled, semi-finished Iodine-133 (isotope) secondary 8 35e-03	2.17e-08
Steel Prod. cold-rolled semi-finished Manganese-54 (isotope) secondary 6 26e-03	1.63e-08
Steel Prod. cold-rolled, semi-finished Iodine-129 (isotope) secondary 6 14e-03	1.59e-08
Invar Palladium-107 (isotope) secondary 5.98e-03	1.55e-08

Process Group	Material	LCI Data Type	Radioactivity (Bq)	% of Total
Ferrite mfg.	Palladium-107 (isotope)	secondary	5.84e-03	1.52e-08
Steel Prod., cold-rolled, semi-finished	Radioactive substance (unspecified)	secondary	3.20e-03	8.31e-09
Invar	Iodine-131 (isotope)	secondary	2.56e-03	6.64e-09
Ferrite mfg.	Iodine-131 (isotope)	secondary	2.49e-03	6.48e-09
Invar	Iodine-133 (isotope)	secondary	2.36e-03	6.14e-09
Ferrite mfg.	Iodine-133 (isotope)	secondary	2.31e-03	5.99e-09
Invar	Manganese-54 (isotope)	secondary	1.77e-03	4.60e-09
Invar	Iodine-129 (isotope)	secondary	1.74e-03	4.52e-09
Ferrite mfg.	Manganese-54 (isotope)	secondary	1.73e-03	4.49e-09
Ferrite mfg.	Iodine-129 (isotope)	secondary	1.70e-03	4.41e-09
Total Materials Processing			3.80e+07	9.88e+01
Manufacturing Life-cycle Stage				0.00e+00
Japanese Electric Grid	Xenon-133M (isotope)	model/secondary	1.96e+04	5.08e-02
US electric grid	Xenon-133 (isotope)	model/secondary	4.98e+03	1.29e-02
Japanese Electric Grid	Tritium-3 (isotope)	model/secondary	2.35e+03	6.10e-03
Japanese Electric Grid	Krypton-85 (isotope)	model/secondary	1.66e+03	4.32e-03
Japanese Electric Grid	Xenon-133 (isotope)	model/secondary	1.30e+03	3.38e-03
Japanese Electric Grid	Argon-41 (isotope)	model/secondary	1.00e+03	2.60e-03
LPG Production	Radioactive substance (unspecified)	secondary	9.21e+02	2.39e-03
Japanese Electric Grid	Xenon-135 (isotope)	model/secondary	7.39e+02	1.92e-03
US electric grid	Tritium-3 (isotope)	model/secondary	5.98e+02	1.55e-03
US electric grid	Krypton-85 (isotope)	model/secondary	4.23e+02	1.10e-03
US electric grid	Xenon-133M (isotope)	model/secondary	3.31e+02	8.59e-04
US electric grid	Argon-41 (isotope)	model/secondary	2.55e+02	6.62e-04
US electric grid	Xenon-135 (isotope)	model/secondary	1.88e+02	4.88e-04
Japanese Electric Grid	Krypton-88 (isotope)	model/secondary	1.41e+02	3.65e-04
Japanese Electric Grid	Xenon-131M (isotope)	model/secondary	1.36e+02	3.52e-04
Japanese Electric Grid	Krypton-85M (isotope)	model/secondary	8.06e+01	2.09e-04
Japanese Electric Grid	Iodine-133 (isotope)	model/secondary	7.03e+01	1.83e-04
US electric grid	Cobalt-58 (isotope)	model/secondary	5.49e+01	1.42e-04
Japanese Electric Grid	Xenon-138 (isotope)	model/secondary	4.68e+01	1.21e-04
US electric grid	Krypton-88 (isotope)	model/secondary	3.58e+01	9.29e-05
US electric grid	Xenon-131M (isotope)	model/secondary	3.45e+01	8.95e-05
Japanese Electric Grid	Krypton-87 (isotope)	model/secondary	3.00e+01	7.79e-05
US electric grid	Krypton-85M (isotope)	model/secondary	2.05e+01	5.32e-05
US electric grid	Iodine-133 (isotope)	model/secondary	1.79e+01	4.65e-05
Japanese Electric Grid	Xenon-135M (isotope)	model/secondary	1.41e+01	3.66e-05
US electric grid	Xenon-138 (isotope)	model/secondary	1.19e+01	3.09e-05
US electric grid	Krypton-87 (isotope)	model/secondary	7.62e+00	1.98e-05
US electric grid	Xenon-135M (isotope)	model/secondary	3.59e+00	9.32e-06
Fuel Oil #6 Prod.	Radioactive substance (unspecified)	secondary	3.12e+00	8.09e-06
Fuel Oil #2 Prod.	Radioactive substance (unspecified)	secondary	2.59e+00	6.71e-06
Japanese Electric Grid	Rubidium-88 (isotope)	model/secondary	3.29e-01	8.55e-07
Fuel Oil #4 Prod.	Radioactive substance (unspecified)	secondary	2.10e-01	5.46e-07
US electric grid	Rubidium-88 (isotope)	model/secondary	8.37e-02	2.17e-07
Japanese Electric Grid	Iodine-134 (isotope)	model/secondary	7.98e-02	2.07e-07
Japanese Electric Grid	Iodine-131 (isotope)	model/secondary	7.58e-02	1.97e-07
Japanese Electric Grid	Chromium-51 (isotope)	model/secondary	6.29e-02	1.63e-07
Japanese Electric Grid	Cesium-137 (isotope)	model/secondary	2.40e-02	6.24e-08
US electric grid	Iodine-134 (isotope)	model/secondary	2.03e-02	5.27e-08
US electric grid	Iodine-131 (isotope)	model/secondary	1.93e-02	5.01e-08

Process Group	Material	LCI Data Type	Radioactivity (Bq) % of	f Total
Natural Gas Prod.	Radioactive substance (unspecified)	secondary	1.86e-02	4.84e-08
Japanese Electric Grid	Cobalt-60 (isotope)	model/secondary	1.62e-02	4.21e-08
US electric grid	Chromium-51 (isotope)	model/secondary	1.60e-02	4.15e-08
Japanese Electric Grid	Iodine-132 (isotope)	model/secondary	1.54e-02	4.00e-08
US electric grid	Cesium-137 (isotope)	model/secondary	6.11e-03	1.59e-08
US electric grid	Cobalt-60 (isotope)	model/secondary	4.13e-03	1.07e-08
Japanese Electric Grid	Iodine-135 (isotope)	model/secondary	4.01e-03	1.04e-08
US electric grid	Iodine-132 (isotope)	model/secondary	3.92e-03	1.02e-08
Japanese Electric Grid	Cesium-134 (isotope)	model/secondary	3.18e-03	8.27e-09
Japanese Electric Grid	Cobalt-58 (isotope)	model/secondary	2.16e-03	5.60e-09
US electric grid	Iodine-135 (isotope)	model/secondary	1.02e-03	2.65e-09
Japanese Electric Grid	Manganese-54 (isotope)	model/secondary	8.92e-04	2.32e-09
US electric grid	Cesium-134 (isotope)	model/secondary	8.09e-04	2.10e-09
US electric grid	Manganese-54 (isotope)	model/secondary	2.27e-04	5.89e-10
Japanese Electric Grid	Cobalt-57 (isotope)	model/secondary	1.69e-04	4.39e-10
Japanese Electric Grid	Bromine-89 (isotope)	model/secondary	1.16e-04	3.01e-10
Japanese Electric Grid	Zirconium-95 (isotope)	model/secondary	9.16e-05	2.38e-10
Japanese Electric Grid	Bromine-90 (isotope)	model/secondary	4.72e-05	1.22e-10
US electric grid	Cobalt-57 (isotope)	model/secondary	4.30e-05	1.12e-10
Japanese Electric Grid	Niobium-95 (isotope)	model/secondary	3.54e-05	9.20e-11
US electric grid	Bromine-89 (isotope)	model/secondary	2.95e-05	7.66e-11
US electric grid	Zirconium-95 (isotope)	model/secondary	2.33e-05	6.05e-11
US electric grid	Bromine-90 (isotope)	model/secondary	1.20e-05	3.11e-11
US electric grid	Niobium-95 (isotope)	model/secondary	9.01e-06	2.34e-11
Japanese Electric Grid	Technetium-99M (isotope)	model/secondary	4.75e-06	1.23e-11
US electric grid	Technetium-99M (isotope)	model/secondary	1.21e-06	3.14e-12
Japanese Electric Grid	Silver-110M (isotope)	model/secondary	1.06e-06	2.75e-12
US electric grid	Silver-110M (isotope)	model/secondary	2.69e-07	6.98e-13
Total Manufacturing			3 500+04	9 100-02
Use Meintenence and Peneir I	ifa avala Staga		5.500104	J.10C-02
Use, Maintenance and Kepan L	Vanan 122 (isatana)	madal/aaaandami	2 12 - 105	8 10 ₂ 01
US electric grid	Tritium 2 (isotope)	model/secondary	3.120+03	0.720.02
US electric grid	Visition 85 (isotope)	model/secondary	2.65 - 104	9.726-02
US electric grid	Krypton-85 (Isotope)	model/secondary	2.03e+04	5.28° 02
US electric grid	Argon 41 (isotope)	model/secondary	2.070+04	3.36e-02
US electric grid	Argon-41 (Isotope)	model/secondary	1.18-+04	4.156-02
US electric grid	Cabalt 58 (isotope)	model/secondary	2.440+02	2.00e-02
US electric grid	Komman 88 (isotope)	model/secondary	2.24=+02	5.936-03
US electric grid	Krypton-88 (Isotope)	model/secondary	2.240+03	5.626-05
US electric grid	Kennten 25M (isotope)	model/secondary	2.100+03	2.22- 02
US electric grid	Krypton-85M (isotope)	model/secondary	1.280+03	2.0102
		model/secondary	1.12e+03	2.916-03
US electric grid	Xenon-138 (isotope)	model/secondary	7.45e+02	1.93e-03
	Krypton-87 (Isotope)	model/secondary	4.780+02	1.24e-03
US electric grid	Derbidium 28 (instead)	model/secondary	2.25e+02	5.84e-04
US electric grid	Kublalum-88 (ISOTOPE)	model/secondary	5.25e+00	1.366-05
US electric grid	Iodine-134 (Isotope)	model/secondary	1.2/e+00	3.30e-06
	roune-131 (Isotope)	model/secondary	1.21e+00	3.14e-06
	Chromium-51 (isotope)	model/secondary	1.00e+00	2.60e-06
	Cesium-137 (isotope)	model/secondary	3.83e-01	9.93e-07
US electric grid	Cobalt-60 (Isotope)	model/secondary	2.59e-01	6./2e-07

 Table M-29. CRT LCIA Results for the Radioactivity Impact Category

Process Group	Material	LCI Data Type	Radioactivity (Bq) %	of Total
US electric grid	Iodine-132 (isotope)	model/secondary	2.45e-01	6.37e-07
US electric grid	Iodine-135 (isotope)	model/secondary	6.39e-02	1.66e-07
US electric grid	Cesium-134 (isotope)	model/secondary	5.07e-02	1.32e-07
US electric grid	Manganese-54 (isotope)	model/secondary	1 42e-02	3 69e-08
US electric grid	Cobalt-57 (isotope)	model/secondary	2.69e-03	6 99e-09
US electric grid	Bromine-89 (isotope)	model/secondary	1.85e-03	4 80e-09
US electric grid	Zirconium-95 (isotope)	model/secondary	1.46e-03	3 79e-09
US electric grid	Bromine-90 (isotope)	model/secondary	7 51e-04	1.95e-09
US electric grid	Niohium-95 (isotope)	model/secondary	5.65e-04	1 47e-09
US electric grid	Technetium-99M (isotone)	model/secondary	7 57e-05	1.97e-10
US electric grid	Silver-110M (isotope)	model/secondary	1.68e-05	4 37e-11
Total Use Maintonance and Der	sair	model/secondary	1.000 05	1 12-100
Total Use, Maintenance and Rep	Jair		4.300+05	1.13e+00
End-of-life Manufacturing Stage	2			
US electric grid	Xenon-133 (isotope)	model/secondary	3.12e+01	8.10e-05
US electric grid	Tritium-3 (isotope)	model/secondary	3.75e+00	9.73e-06
US electric grid	Krypton-85 (isotope)	model/secondary	2.65e+00	6.88e-06
US electric grid	Xenon-133M (isotope)	model/secondary	2.07e+00	5.38e-06
US electric grid	Argon-41 (isotope)	model/secondary	1.60e+00	4.15e-06
US electric grid	Xenon-135 (isotope)	model/secondary	1.18e+00	3.06e-06
US electric grid	Cobalt-58 (isotope)	model/secondary	3.44e-01	8.93e-07
US electric grid	Krypton-88 (isotope)	model/secondary	2.24e-01	5.82e-07
US electric grid	Xenon-131M (isotope)	model/secondary	2.16e-01	5.61e-07
US electric grid	Krypton-85M (isotope)	model/secondary	1.28e-01	3.33e-07
US electric grid	Iodine-133 (isotope)	model/secondary	1.12e-01	2.91e-07
US electric grid	Xenon-138 (isotope)	model/secondary	7.45e-02	1.94e-07
US electric grid	Krypton-87 (isotope)	model/secondary	4.78e-02	1.24e-07
US electric grid	Xenon-135M (isotope)	model/secondary	2.25e-02	5.84e-08
LPG Production	Radioactive substance (unspecified)	secondary	7.95e-03	2.07e-08
US electric grid	Rubidium-88 (isotope)	model/secondary	5.25e-04	1.36e-09
US electric grid	Iodine-134 (isotope)	model/secondary	1.27e-04	3.30e-10
US electric grid	Iodine-131 (isotope)	model/secondary	1.21e-04	3.14e-10
US electric grid	Chromium-51 (isotope)	model/secondary	1.00e-04	2.60e-10
US electric grid	Cesium-137 (isotope)	model/secondary	3.83e-05	9.94e-11
US electric grid	Cobalt-60 (isotope)	model/secondary	2.59e-05	6.72e-11
US electric grid	Iodine-132 (isotope)	model/secondary	2.46e-05	6.38e-11
US electric grid	Iodine-135 (isotope)	model/secondary	6.39e-06	1.66e-11
US electric grid	Cesium-134 (isotope)	model/secondary	5.07e-06	1.32e-11
US electric grid	Manganese-54 (isotope)	model/secondary	1.42e-06	3.69e-12
US electric grid	Cobalt-57 (isotope)	model/secondary	2.69e-07	7.00e-13
US electric grid	Bromine-89 (isotope)	model/secondary	1.85e-07	4.80e-13
US electric grid	Zirconium-95 (isotope)	model/secondary	1.46e-07	3.79e-13
US electric grid	Bromine-90 (isotope)	model/secondary	7.52e-08	1.95e-13
US electric grid	Niobium-95 (isotope)	model/secondary	5.65e-08	1.47e-13
US electric grid	Technetium-99M (isotope)	model/secondary	7.58e-09	1.97e-14
US electric grid	Silver-110M (isotope)	model/secondary	1.68e-09	4.38e-15
Natural Gas Prod.	Radioactive substance (unspecified)	secondary	-9.34e-03	-2.43e-08
CRT Incineration	Radioactive substance (unspecified)	secondary	-1.45e-01	-3.76e-07
Fuel Oil #4 Prod.	Radioactive substance (unspecified)	secondary	-2.26e+00	-5.87e-06
Total End-of-life		-	4.12e+01	1.07e-04
Total All Life-cycle Stages			3 850+07	1 00e+02
			5.050.07	1.000.02

Process Group	Material	LCI Data Type	Radioactivity (Bg)	% of Total
Materials Processing Life-cycle Sta	lge			
Steel Prod cold-rolled semi-finished	Plutonium-241 (isotone)	secondary	1 18e+07	9.64e+01
Steel Prod., cold-rolled, semi-finished	Plutonium-240 (isotope)	secondary	5.08e+04	4 16e-01
Steel Prod., cold-rolled, semi-finished	Cesium-135 (isotope)	secondary	4 59e+04	3.76e-01
Steel Prod. cold-rolled semi-finished	Radon-222 (isotope)	secondary	4.30e+04	3.52e-01
Steel Prod., cold-rolled, semi-finished	Plutonium-239 (isotope)	secondary	3.57e+04	2.93e-01
Steel Prod., cold-rolled, semi-finished	Strontium-90 (isotope)	secondary	6.86e+03	5.62e-02
Steel Prod. cold-rolled semi-finished	Tritium-3 (isotope)	secondary	1 23e+03	1.00e-02
Steel Prod., cold-rolled, semi-finished	radioactive gas (unspecified)	secondary	8.98e+03	7 36e-03
Steel Prod., cold rolled, semi-finished	Venon 133 (isotone)	secondary	7.63e+02	6 25e 03
Steel Prod., cold-rolled, semi-finished	Radium-226 (isotope)	secondary	4.24e+02	3.48e-03
Steel Prod., cold rolled, semi-finished	Thorium 230 (isotope)	secondary	2.53e+02	2.07e.03
Steel Prod., cold-rolled, semi-finished	Plutonium 242 (isotope)	secondary	1.020+02	1.570.03
Steel Prod., cold rolled, semi-finished	Curium 244 (isotope)	secondary	1.920+02	1.57e.03
Steel Frod., cold-tolled, semi-finished	Liranium 224 (isotope)	secondary	1.910+02	1.376-03
Steel Prod., cold-rolled, semi-finished	A maniaium 241 (isotope)	secondary	0.42a+01	7.722.04
Steel Prod., cold-rolled, semi-finished	Americium-241 (isotope)	secondary	9.42e+01	1.72e-04
Steel Prod., cold-rolled, semi-finished	Krypton-85 (isotope)	secondary	3.430+01	4.476-04
Steel Prod., cold-rolled, semi-linished	Granium-238 (isotope)	secondary	4.55e+01	3.72e-04
Steel Prod., cold-rolled, semi-finished	Samarium-151 (isotope)	secondary	4.25e+01	3.48e-04
Steel Prod., cold-rolled, semi-finished	Neptunium-237 (isotope)	secondary	2.95e+01	2.42e-04
Steel Prod., cold-rolled, semi-finished	Carbon-14 (isotope)	secondary	9.35e+00	7.666-05
Steel Prod., cold-rolled, semi-finished	Radium-222 (isotope)	secondary	3.26e+00	2.6/e-05
Steel Prod., cold-rolled, semi-finished	Uranium-235 (isotope)	secondary	2.87e+00	2.35e-05
Steel Prod., cold-rolled, semi-finished	Americium-243 (isotope)	secondary	2.05e+00	1.68e-05
Steel Prod., cold-rolled, semi-finished	Technetium-99M (isotope)	secondary	1.40e+00	1.15e-05
Natural Gas Prod.	Radioactive substance (unspecified)	secondary	1.30e+00	1.07e-05
Steel Prod., cold-rolled, semi-finished	Thorium-228 (isotope)	secondary	1.22e+00	1.00e-05
Steel Prod., cold-rolled, semi-finished	Radon-220 (isotope)	secondary	9.93e-01	8.13e-06
Steel Prod., cold-rolled, semi-finished	Radium-228 (isotope)	secondary	6.36e-01	5.21e-06
Steel Prod., cold-rolled, semi-finished	Polonium-210 (isotope)	secondary	5.52e-01	4.52e-06
Steel Prod., cold-rolled, semi-finished	Lead-210 (isotope)	secondary	3.21e-01	2.63e-06
Steel Prod., cold-rolled, semi-finished	Radium-224 (isotope)	secondary	2.97e-01	2.43e-06
Steel Prod., cold-rolled, semi-finished	Zirconium-93 (isotope)	secondary	1.84e-01	1.51e-06
Steel Prod., cold-rolled, semi-finished	Cesium-137 (isotope)	secondary	1.56e-01	1.28e-06
Steel Prod., cold-rolled, semi-finished	Protactinium-234 (isotope)	secondary	9.87e-02	8.09e-07
Steel Prod., cold-rolled, semi-finished	Thorium-234 (isotope)	secondary	9.87e-02	8.09e-07
Steel Prod., cold-rolled, semi-finished	Silver-110M (isotope)	secondary	9.21e-02	7.54e-07
Steel Prod., cold-rolled, semi-finished	Potassium-40 (isotope)	secondary	8.57e-02	7.02e-07
Steel Prod., cold-rolled, semi-finished	Cobalt-58 (isotope)	secondary	6.17e-02	5.06e-07
Steel Prod., cold-rolled, semi-finished	Tin-126 (isotope)	secondary	5.79e-02	4.74e-07
Steel Prod., cold-rolled, semi-finished	Cobalt-60 (isotope)	secondary	3.87e-02	3.17e-07
Steel Prod., cold-rolled, semi-finished	Selenium-79 (isotope)	secondary	3.31e-02	2.71e-07
Steel Prod cold-rolled semi-finished	Radioactive aerosols and halogenes (unspecified)	secondary	2.81e-02	2 30e-07
Steel Prod. cold-rolled semi-finished	Thorium-232 (isotone)	secondary	2.010-02	1.87e-07
Steel Prod. cold-rolled semi-finished	Curium-245 (isotope)	secondary	2.230-02	1.070-07
Steel Prod. cold-rolled semi-finished	Antimony-124 (isotope)	secondary	2.130-02	1.750-07
Steel Prod. cold-rolled semi-finished	Cesium-134 (isotope)	secondary	1 900-02	1.56e-07
Steel Prod. cold-rolled semi-finished	Palladium-107 (isotope)	secondary	1.04e-02	8 48e-08
and a solution of the solution	- Shuanani I Si (ISOtope)	Secondary	1.010-02	0.100 00

Table M-30. LCD LCIA Results for the Radioactivity Impact Category

APPENDIX M

Steel Prod., cold-rolled, semi-finishedIodine-131 (isotope)secondary4.42e-033.62e-0Steel Prod., cold-rolled, semi-finishedIodine-133 (isotope)secondary4.09e-033.35e-0Steel Prod., cold-rolled, semi-finishedManganese-54 (isotope)secondary3.07e-032.51e-0
Steel Prod., cold-rolled, semi-finishedIodine-133 (isotope)secondary4.09e-033.35e-0Steel Prod., cold-rolled, semi-finishedManganese-54 (isotope)secondary3.07e-032.51e-0
Steel Prod. cold-rolled semi-finished Manganese-54 (isotone) secondary 3.07a.03 2.51a.0
5001100, 5001000 , 5001000 500000 5000000 5000000 50000000 50000000000
Steel Prod., cold-rolled, semi-finished Iodine-129 (isotope) secondary 3.01e-03 2.47e-0
Steel Prod., cold-rolled, semi-finished Radioactive substance (unspecified) secondary 1.57e-03 1.28e-0
Total Materials Processing 1 20e+07 9 79e+0
Manufacturing Life avala Stage
Internação Electrico Crid Venero 122M (instano) model/accordore: 659a104 520a.0
Japanese Electric Orid Xenon-155M (Isotope) model/secondary 0.58e+04 5.59e-0
Japanese Electric Grid Innum-5 (Isotope) model/secondary 7.90e+05 6.48e-0.
Japanese Electric Grid Krypton-85 (Isotope) model/secondary 5.596+05 4.586-0.
Japanese Electric Orid Arean 41 (jostene) model/secondary 4.57e+05 5.58e-0.
Japanese Electric Orid Argon-41 (isotope) model/secondary 5.57e+05 2.70e-0.
Japanese Electric Grid Xenon-135 (isotope) model/secondary 2.48e+05 2.04e-0.
Use rectificing and Action 135 (isotope) indef/secondary 0.04c+02 4.95c-0.
Japanese Electric Orid Krypton-88 (isotope) indee/secondary 4.75e+02 5.88e-0.
Japanese Electric Orid Xenon-151M (isotope) model/secondary 4.50e+02 5.74e-0.
Japanese Electric Orid Krypton-85M (Isotope) model/secondary 2.71e+02 2.22e-0.
Japanese Electric Orid Tourie-155 (isotope) model/secondary 2.57e+02 1.94e-0.
Japanese Electric Orid Xenon-138 (isotope) model/secondary 1.57e+02 1.29e-0.
Japanese Electric Grid Krypton-87 (Isotope) model/secondary 1.01e+02 8.26e-0
US electric grid Infiniti-S (isotope) indel/secondary 7.25e+01 5.94e-0
Userence grid Krypton-85 (isotope) indel/secondary 5.15e+01 4.20e-0
Japanese Electric Orid Action-155M (Isotope) Inode/secondary 4.74±+01 5.89±-0 L DG Production Padioactive substance (unspecified) secondary 4.42±+01 3.62±-0
Let O Froduction Radioactive substance (unspectified) secondary 4.42c+01 5.02c-0.
US electric grid Argon 41 (isotope) model/secondary $4.01c+01$ $5.29c-0$
US electric grid Xenon 135 (isotope) model/secondary 2.28+01 1.87-0
US electric grid Cobalt 58 (isotope) model/secondary 2.200-01 1.870-0
US electric grid $V_{\text{subtrans}} = \frac{V_{\text{subtrans}}}{V_{\text{subtrans}}} = \frac{V_{\text{subtrans}}}{V_{\text{subtrans}$
US electric grid Venon 131M (isotope) model/secondary 4.546+00 3.43e 0
US electric grid Krynton 85M (isotope) model/secondary 2.48+00 2.04e 0
US electric grid Isotope) model/secondary 2.460-00 2.040-0
US electric grid Venon-138 (isotope) model/secondary 2.170-00 1.18e-0
Japanese Electric Grid Rubidium-88 (isotope) model/secondary 1.11e+00 9.07e-0
US electric grid Kriviton 87 (jsotone) model/secondary 0.25e 01 7.58e 0
US electric grid Venon-135M (isotope) model/secondary 9.250-01 7.560-0
Fuel Oil #4 Prod Radioactive substance (unspecified) secondary 3.24e-01 2.65e-0
Inter on #4 from Radioactive substance (unspectified) secondary 5.240-01 2.050-0 Japanese Electric Grid Jodine-134 (isotope) model/secondary 2.68e-01 2.20e-0
Japanese Electric Grid Iodine-131 (isotope) model/secondary 2.55e-01 2.20e-0
Japanese Electric Grid Chromium-51 (isotope) model/secondary 2.15e 01 2.09e 0
Fuel Oil #2 Prod Radioactive substance (unspecified) secondary 1.21e-01 9.88e-0
Fuel Oil #6 Prod Radioactive substance (unspecified) secondary 1.06e-01 8 69e-0
Indef of worked Indef of worked
Japanese Electric Grid Cobalt-60 (isotope) model/secondary 546e-02 4 47e-0
Japanese Electric Grid Iodine-132 (isotope) model/secondary 5.166-02 4.24e-0
Natural Gas Prod Radioactive substance (unspecified) secondary 294e-02 241e-0
Japanese Electric Grid Iodine-135 (isotope) model/secondary 1.35e-02 1.10e-0
Japanese Electric Grid Cesium-134 (isotope) model/secondary 1.056-02 8.77e-0
US electric grid Rubidium-88 (isotope) model/secondary 1.02e-02 8.32e-0
Japanese Electric Grid Cobalt-58 (isotope) model/secondary 7 26e-03 5 94e-0
Japanese Electric Grid Manganese-54 (isotope) model/secondary 3.00e-03 2.46e-0

Process Group	Material	LCI Data Type	Radioactivity (Bq)	% of Total
US electric grid	Iodine-134 (isotope)	model/secondary	2.46e-03	2.02e-08
US electric grid	Iodine-131 (isotope)	model/secondary	2.34e-03	1.92e-08
US electric grid	Chromium-51 (isotope)	model/secondary	1.94e-03	1.59e-08
US electric grid	Cesium-137 (isotope)	model/secondary	7.41e-04	6.07e-09
Japanese Electric Grid	Cobalt-57 (isotope)	model/secondary	5.68e-04	4.66e-09
US electric grid	Cobalt-60 (isotope)	model/secondary	5.01e-04	4.10e-09
US electric grid	Iodine-132 (isotope)	model/secondary	4.75e-04	3.89e-09
Japanese Electric Grid	Bromine-89 (isotope)	model/secondary	3.90e-04	3.19e-09
Japanese Electric Grid	Zirconium-95 (isotope)	model/secondary	3.08e-04	2.52e-09
Japanese Electric Grid	Bromine-90 (isotope)	model/secondary	1.59e-04	1.30e-09
US electric grid	Iodine-135 (isotope)	model/secondary	1.24e-04	1.01e-09
Japanese Electric Grid	Niobium-95 (isotope)	model/secondary	1.19e-04	9.77e-10
US electric grid	Cesium-134 (isotope)	model/secondary	9.82e-05	8.04e-10
US electric grid	Manganese-54 (isotope)	model/secondary	2.75e-05	2.25e-10
Japanese Electric Grid	Technetium-99M (isotope)	model/secondary	1.60e-05	1.31e-10
US electric grid	Cobalt-57 (isotope)	model/secondary	5.21e-06	4.27e-11
US electric grid	Bromine-89 (isotope)	model/secondary	3.58e-06	2.93e-11
Japanese Electric Grid	Silver-110M (isotope)	model/secondary	3.56e-06	2.91e-11
US electric grid	Zirconium-95 (isotope)	model/secondary	2.82e-06	2.31e-11
US electric grid	Bromine-90 (isotope)	model/secondary	1.45e-06	1.19e-11
US electric grid	Niobium-95 (isotope)	model/secondary	1.09e-06	8.96e-12
US electric grid	Technetium-99M (isotope)	model/secondary	1.47e-07	1.20e-12
US electric grid	Silver-110M (isotope)	model/secondary	3.26e-08	2.67e-13
Total Manufacturing			9.22e+04	7.55e-01
Use, Maintenance and Repair Life-o	cycle Stage			
US electric grid	Xenon-133 (isotope)	model/secondary	1.16e+05	9.53e-01
US electric grid	Tritium-3 (isotope)	model/secondary	1.40e+04	1.14e-01
US electric grid	Krypton-85 (isotope)	model/secondary	9.88e+03	8.10e-02
US electric grid	Xenon-133M (isotope)	model/secondary	7.73e+03	6.34e-02
US electric grid	Argon-41 (isotope)	model/secondary	5.96e+03	4.88e-02
US electric grid	Xenon-135 (isotope)	model/secondary	4.39e+03	3.60e-02
US electric grid	Cobalt-58 (isotope)	model/secondary	1.28e+03	1.05e-02
US electric grid	Krypton-88 (isotope)	model/secondary	8.37e+02	6.85e-03
US electric grid	Xenon-131M (isotope)	model/secondary	8.06e+02	6.60e-03
US electric grid	Krypton-85M (isotope)	model/secondary	4.79e+02	3.92e-03
US electric grid	Iodine-133 (isotope)	model/secondary	4.18e+02	3.43e-03
US electric grid	Xenon-138 (isotope)	model/secondary	2.78e+02	2.28e-03
US electric grid	Krypton-87 (isotope)	model/secondary	1.78e+02	1.46e-03
US electric grid	Xenon-135M (isotope)	model/secondary	8.39e+01	6.87e-04
US electric grid	Rubidium-88 (isotope)	model/secondary	1.96e+00	1.60e-05
US electric grid	Iodine-134 (isotope)	model/secondary	4.74e-01	3.89e-06
US electric grid	Iodine-131 (isotope)	model/secondary	4.51e-01	3.69e-06
US electric grid	Chromium-51 (isotope)	model/secondary	3.74e-01	3.06e-06
US electric grid	Cesium-137 (isotope)	model/secondary	1 43e-01	1 17e-06
US electric grid	Cobalt-60 (isotope)	model/secondary	9.65e-02	7.91e-07
US electric grid	Iodine-132 (isotope)	model/secondary	9.16e-02	7.50e-07
US electric grid	Iodine-135 (isotope)	model/secondary	2.38e-02	1 95e-07
US electric grid	Cesium-134 (isotope)	model/secondary	1.89e-02	1.55e-07
US electric grid	Manganese-54 (isotope)	model/secondary	5.30e-03	4.35e-08
0	- · · · · · · · · · · · · · · · · · · ·			

Table M-30. LCD LCIA Results for the Radioactivity Impact Category

Process Group	Material	LCI Data Type	Radioactivity (Bq)	% of Total
US electric grid	Cobalt-57 (isotope)	model/secondary	1.00e-03	8.23e-09
US electric grid	Bromine-89 (isotope)	model/secondary	6.89e-04	5.65e-09
US electric grid	Zirconium-95 (isotope)	model/secondary	5.44e-04	4.46e-09
US electric grid	Bromine-90 (isotope)	model/secondary	2.80e-04	2.30e-09
US electric grid	Niobium-95 (isotope)	model/secondary	2.11e-04	1.73e-09
US electric grid	Technetium-99M (isotope)	model/secondary	2.83e-05	2.32e-10
US electric grid	Silver-110M (isotope)	model/secondary	6.29e-06	5.15e-11
Total Use, Maintenance and Repair			1.63e+05	1.33e+00
End-of-life Life-cycle Stage				
US electric grid	Xenon-133 (isotope)	model/secondary	2.21e+01	1.81e-04
US electric grid	Tritium-3 (isotope)	model/secondary	2.65e+00	2.17e-05
US electric grid	Krypton-85 (isotope)	model/secondary	1.88e+00	1.54e-05
US electric grid	Xenon-133M (isotope)	model/secondary	1.47e+00	1.20e-05
US electric grid	Argon-41 (isotope)	model/secondary	1.13e+00	9.26e-06
US electric grid	Xenon-135 (isotope)	model/secondary	8.33e-01	6.83e-06
US electric grid	Cobalt-58 (isotope)	model/secondary	2.43e-01	1.99e-06
US electric grid	Krypton-88 (isotope)	model/secondary	1.59e-01	1.30e-06
US electric grid	Xenon-131M (isotope)	model/secondary	1.53e-01	1.25e-06
US electric grid	Krypton-85M (isotope)	model/secondary	9.09e-02	7.45e-07
US electric grid	Iodine-133 (isotope)	model/secondary	7.94e-02	6.50e-07
US electric grid	Xenon-138 (isotope)	model/secondary	5.28e-02	4.32e-07
US electric grid	Krypton-87 (isotope)	model/secondary	3.38e-02	2.77e-07
US electric grid	Xenon-135M (isotope)	model/secondary	1.59e-02	1.30e-07
LPG Production	Radioactive substance (unspecified)	secondary	3.63e-03	2.97e-08
US electric grid	Rubidium-88 (isotope)	model/secondary	3.72e-04	3.04e-09
US electric grid	Iodine-134 (isotope)	model/secondary	9.00e-05	7.37e-10
US electric grid	Iodine-131 (isotope)	model/secondary	8.56e-05	7.01e-10
US electric grid	Chromium-51 (isotope)	model/secondary	7.09e-05	5.81e-10
US electric grid	Cesium-137 (isotope)	model/secondary	2.71e-05	2.22e-10
US electric grid	Cobalt-60 (isotope)	model/secondary	1.83e-05	1.50e-10
US electric grid	Iodine-132 (isotope)	model/secondary	1.74e-05	1.42e-10
US electric grid	Iodine-135 (isotope)	model/secondary	4.52e-06	3.70e-11
US electric grid	Cesium-134 (isotope)	model/secondary	3.59e-06	2.94e-11
US electric grid	Manganese-54 (isotope)	model/secondary	1.01e-06	8.25e-12
US electric grid	Cobalt-57 (isotope)	model/secondary	1.91e-07	1.56e-12
US electric grid	Bromine-89 (isotope)	model/secondary	1.31e-07	1.07e-12
US electric grid	Zirconium-95 (isotope)	model/secondary	1.03e-07	8.46e-13
US electric grid	Bromine-90 (isotope)	model/secondary	5.32e-08	4.36e-13
US electric grid	Niobium-95 (isotope)	model/secondary	4.00e-08	3.28e-13
US electric grid	Technetium-99M (isotope)	model/secondary	5.36e-09	4.39e-14
US electric grid	Silver-110M (isotope)	model/secondary	1.19e-09	9.77e-15
Natural Gas Prod.	Radioactive substance (unspecified)	secondary	-6.16e-03	-5.05e-10
LCD incineration	Radioactive substance (unspecified)	secondary	-9.39e-02	-7.69e-09
Fuel Oil #4 Prod.	Radioactive substance (unspecified)	secondary	-1.49e+00	-1.22e-07
Total End-of-life			2.93e+01	2.53e-04
Total All Life-cycle Stages			1.22e+07	1.00e+02

9.31e-03

9.96e-04

Process Group Material LCI Data Type **Chronic Occupational** % of Total Toxicity (tox-kg) Materials Processing Life-cycle Stage 7.74e-02 Steel Prod., cold-rolled, semi-finished Sodium Dichromate 7.23e-01 secondary 6.68e-04 Steel Prod., cold-rolled, semi-finished Phenolsulphonic Acid secondary 6.24e-03 2.44e-04 Aluminum Prod. Barium sulfate 2.28e-03 secondary Steel Prod., cold-rolled, semi-finished secondary 2.46e-04 2.64e-05 Dioctyl Sebacate 2.29e-05 2.14e-04 Ferrite mfg. Barium sulfate secondary Steel Prod., cold-rolled, semi-finished Etoxy Naphtol Sulphonic Acid (ENSA) secondary 1.99e-04 2.13e-05 Steel Prod., cold-rolled, semi-finished Barium sulfate 1.90e-04 2.03e-05 secondary **ABS** Production Barium sulfate 9.07e-08 secondary 8.47e-07 Borax 5.26e-07 5.64e-08 Invar secondary 5.50e-08 Ferrite mfg. Borax secondary 5.14e-07 **Total Materials Processing** 7.32e-01 7.84e-02 Manufacturing Life-cycle Stage Glass/frit Liquified petroleum gas ("propane") 7.02e+02 7.51e+01 primary PWB Mfg. Sulfuric acid model/secondary 1.26e+02 1.34e+01 CRT tube mfg. Sulfuric acid 3.84e+01 4.11e+00 primary Glass/frit Barium Carbonate primary 1.68e+01 1.80e+00CRT tube mfg. Fuel oil #6 primary 7.37e+00 7.89e-01 Glass/frit Fuel oil #2 2.32e+00 2.49e-01 primary PPE 1.57e-01 CRT monitor assembly 1.47e+00 primary Glass/frit Lead primary 8.95e-01 9.58e-02 PWB Mfg. Hydrochloric acid 9.40e-02 model/secondary 8.78e-01 PWB Mfg. Formaldehyde model/secondary 7.56e-01 8.09e-02 CRT tube mfg. LNG 6.71e-01 7.18e-02 primary CRT tube mfg. Ammonium bifluoride 4.85e-01 5.20e-02 primary 4.86e-02 Glass/frit. Cerium Oxide 4.53e-01 primary Glass/frit Strontium Carbonate 3.52e-01 3.77e-02 primary Glass/frit Sodium Dichromate 2.88e-01 3.09e-02 primary CRT monitor assembly Fuel oil #4 primary 2.74e-01 2.93e-02 PWB Mfg. Nitric acid model/secondary 2.72e-01 2.91e-02 CRT tube mfg. Hydrochloric acid primary 2.01e-01 2.15e-02 1.69e-02 Glass/frit Hydrofluoric acid 1.58e-01 primary 1.66e-02 PWB Mfg. Ammonium hydroxide model/secondary 1.55e-01 CRT tube mfg. Sulfuric acid, aluminum salt primary 1.28e-02 1.20e-01 Glass/frit Zircon Sand primary 1.09e-01 1.16e-02 CRT tube mfg. Hydrogen peroxide 1.07e-01 1.14e-02 primary CRT monitor assembly Triphenyl phosphate primary 1.06e-01 1.13e-02 Glass/frit Lead 9.33e-02 9.99e-03 primary CRT tube mfg. Chlorine primary 8.07e-02 8.64e-03 Aluminum Oxide 6.74e-02 7.22e-03 CRT glass mfg. primary PWB Mfg. Hydrogen peroxide model/secondary 6.21e-02 6.65e-03 CRT monitor assembly Tricresyl phosphate primary 4.59e-02 4.92e-03 CRT monitor assembly Isopropyl alcohol primary 2.44e-02 2.61e-03 CRT tube mfg. Nitric acid 1.63e-02 1.75e-03 primary Glass/frit Borax primary 1.60e-02 1.71e-03 CRT tube mfg. Hydrofluoric acid 1.48e-02 1.58e-03 primary CRT tube mfg. Boric acid primary 1.37e-02 1.47e-03 PWB Mfg. Glycol ethers model/secondary 1.34e-02 1.43e-03

Table M-31. CRT LCIA Results for the Chronic Occupational Health Effects Impact Category

primary

Red Phosphor (Y2O2S)

CRT tube mfg.

Process Group	Material	LCI Data Type	Chronic Occupational Toxicity (tox-kg)	% of Total
CRT monitor assembly	Phosphate ester	primary	8.41e-03	9.00e-04
CRT tube mfg.	Polyvinyl alcohol	primary	8.11e-03	8.69e-04
CRT tube mfg.	Blue Phosphor (ZnS)	primary	7.67e-03	8.22e-04
CRT tube mfg.	Green Phosphor (ZnS)	primary	6.68e-03	7.15e-04
CRT tube mfg.	Chromium (VI)	primary	5.68e-03	6.09e-04
CRT tube mfg.	Sodium Metabisulfite	primary	4.67e-03	5.00e-04
CRT tube mfg.	Dimethyl Formamide	primary	3.84e-03	4.11e-04
CRT tube mfg.	Isopentylacetate	primary	3.49e-03	3.74e-04
CRT tube mfg.	Blue Phosphor (ZnS.Ag.Al)	primary	3.34e-03	3.57e-04
CRT tube mfg.	Ammonium hydroxide	primary	2.81e-03	3.01e-04
CRT tube mfg.	Green Phosphor (ZnS.Cu.Al)	primary	2.68e-03	2.87e-04
CRT tube mfg.	Red Phosphor (Y2O2S.Eu)	primary	2.67e-03	2.86e-04
CRT tube mfg.	Amyl acetate (mixed isomers)	primary	2.41e-03	2.58e-04
CRT tube mfg.	Ammonium fluoride	primary	1.78e-03	1.91e-04
CRT tube mfg.	Toluene	primary	8.02e-04	8.59e-05
CRT tube mfg.	Sodium Persulfate	primary	7.08e-04	7.58e-05
CRT tube mfg.	Ammonium Oxalate Monohydrate	primary	6.32e-04	6.77e-05
Japanese Electric Grid	Uranium, yellowcake	model/secondary	6.07e-04	6.50e-05
CRT tube mfg.	Sodium Hypochlorite	primary	5.24e-04	5.61e-05
CRT tube mfg.	Polyvinyl Pyrrolidone (PVP)	primary	5.22e-04	5.59e-05
CRT tube mfg.	Periodic Acid	primary	4.53e-04	4.85e-05
CRT tube mfg.	Ammonia	primary	3.24e-04	3.47e-05
CRT monitor assembly	2,2,4-trimethylpentane	primary	3.01e-04	3.22e-05
US electric grid	Uranium, yellowcake	model/secondary	1.55e-04	1.66e-05
CRT tube mfg.	Ammonium Dichromate	primary	6.99e-05	7.49e-06
CRT tube mfg.	Sodium Dichromate Dihydrate (VI)	primary	6.19e-05	6.63e-06
CRT tube mfg.	Acetone	primary	3.77e-05	4.03e-06
CRT monitor assembly	Fluorocarbon resin	primary	3.75e-05	4.02e-06
CRT tube mfg.	Xylene (mixed isomers)	primary	3.19e-05	3.42e-06
CRT monitor assembly	Cyclohexane	primary	8.61e-06	9.22e-07
Total Manufacturing			9.00e+02	9.64e+01
Use, Maintenance and Repair Life-o	cycle Stage			
US electric grid	Natural gas	model/secondary	2.80e+01	3.00e+00
US electric grid	Petroleum (in gournd)	model/secondary	7.60e+00	8.14e-01
US electric grid	Uranium, yellowcake	model/secondary	9.70e-03	1.04e-03
Total Use, Maintenance and Repair			3.56e+01	3.81e+00
End-of-life Life-cycle Stage				
CRT landfilling	Fuel oil #4	primary	3.86e-02	4.13e-03
CRT Recycling	Liquified petroleum gas ("propane")	primary	6.06e-03	6.48e-04
US electric grid	Uranium, yellowcake	model/secondary	9.71e-07	1.04e-07
CRT Incineration	Fuel oil #4	secondary	-2.88e+00	-3.08e-01
Total End-of-life			-2.83e+00	-3.03e-01
Total All Life-cycle Stages			9.34e+02	1.00e+02

			Chronic Occupational	
Process Group	Materials	LCI Data Type	Toxicity (tox-kg)	% of Total
Materials Processing Life-cycle Stag	ze	£.		
Steel Prod., cold-rolled, semi-finished	Sodium Dichromate	secondary	3.54e-01	5.09e-02
Steel Prod., cold-rolled, semi-finished	Phenolsulphonic Acid	secondary	3.06e-03	4.40e-04
Aluminum Prod.	Barium sulfate	secondary	8.47e-04	1.22e-04
Steel Prod., cold-rolled, semi-finished	Dioctyl Sebacate	secondary	1.21e-04	1.73e-05
Steel Prod., cold-rolled, semi-finished	Etoxy Naphtol Sulphonic Acid (ENSA)	secondary	9.76e-05	1.40e-05
Steel Prod., cold-rolled, semi-finished	Barium sulfate	secondary	9.30e-05	1.34e-05
PMMA Sheet Prod.	Barium sulfate	secondary	2.42e-05	3.47e-06
PMMA Sheet Prod.	Ferromanganese (Fe, Mn, C)	secondary	7.67e-07	1.10e-07
Total Materials Processing		,	3.59e-01	5.15e-02
Manufacturing Life-cycle Stage			5,550 01	0.100 02
Monitor/module	LNG	nrimary	3 95e+02	5 67e+01
Monitor/module	Sulfuric acid	primary	1 58e+02	2.26e+01
PWB Mfg	Sulfuric acid	model/secondary	5 54e+01	7.96e+00
LCD glass mfg	Liquified petroleum gas ("propane")	primary	3.54e+01	4.67e+00
Monitor/module	Phosphine	primary	1 23e+01	1.77e+00
Panel components	Sulfuric acid	primary	1.09e+01	1.56e+0.0
Monitor/module	Dimethylsulfoxide	primary	7.96e+00	1.14e+00
Monitor/module	Ethanol amine	primary	4 32e+00	6.21e-01
Panel components	Tetrahydrofuran	primary	1 32e+00	1.89e-01
Monitor/module	Liquified petroleum gas ("propane")	primary	1.17e+00	1.67e-01
LCD glass mfg.	Barium Carbonate	primary	7.79e-01	1.12e-01
Monitor/module	Kerosene	primary	5 95e-01	8 56e-02
Monitor/module	Ammonium bifluoride	primary	5.62e-01	8.07e-02
Monitor/module	Fuel oil #4	primary	4.22e-01	6.06e-02
Monitor/module	Isopropyl alcohol	primary	4.19e-01	6.02e-02
PWB Mfg.	Hydrochloric acid	model/secondary	3.87e-01	5.57e-02
Panel components	Kerosene	primary	3.35e-01	4.82e-02
PWB Mfg.	Formaldehvde	model/secondary	3.33e-01	4.79e-02
Monitor/module	Tetramethyl ammonium hydroxide	primary	2.57e-01	3.70e-02
Panel components	Fuel oil #6	primary	2.51e-01	3.61e-02
Monitor/module	Nitrogen fluoride	primary	2.16e-01	3.11e-02
Monitor/module	Hydrochloric acid	primary	1.97e-01	2.83e-02
Monitor/module	Phosphoric acid	primary	1.90e-01	2.73e-02
Monitor/module	2-(2-butoxyethoxy)-ethanol	primary	1.81e-01	2.60e-02
Monitor/module	N-Butylacetate	primary	1.64e-01	2.35e-02
Monitor/module	Molybdenum	primary	1.52e-01	2.18e-02
PWB Mfg.	Nitric acid	model/secondary	1.20e-01	1.72e-02
LCD glass mfg.	Fuel oil #2	primary	1.08e-01	1.55e-02
Monitor/module	Hydrofluoric acid	primary	8.42e-02	1.21e-02
PWB Mfg.	Ammonium hydroxide	model/secondary	6.85e-02	9.84e-03
Monitor/module	Ammonia	primary	4.22e-02	6.06e-03
Monitor/module	Sulfur hexafluoride	primary	3.24e-02	4.66e-03
	Propylene glycol monomethyl ether			
Monitor/module	acetate	primary	3.12e-02	4.49e-03
Monitor/module	Chlorine	primary	3.11e-02	4.46e-03
PWB Mfg.	Hydrogen peroxide	model/secondary	2.74e-02	3.94e-03
Monitor/module	Nitric acid	primary	2.49e-02	3.58e-03
Monitor/module	Ammonium fluoride	primary	2.28e-02	3.28e-03

Table M-32. LCD LCIA Results for the Chronic Occupational Health Effects Impact Category

			Chronic Occupational	
Process Group	Materials	LCI Data Type	Toxicity (tox-kg)	% of Total
LCD glass mfg.	Cerium Oxide	primary	2.10e-02	2 3.02e-03
Monitor/module	Isopropyl alcohol	primary	1.88e-02	2 2.70e-03
Monitor/module	Acetone	primary	1.74e-02	2 2.49e-03
LCD glass mfg.	Strontium Carbonate	primary	1.63e-02	2 2.34e-03
Monitor/module	Glycol ethers	primary	1.23e-02	2 1.76e-03
Monitor/module	1-Methoxy-2-propanol	primary	1.21e-02	2 1.74e-03
Monitor/module	Titanium	primary	1.16e-02	2 1.66e-03
Panel components	Polyvinyl alcohol	primary	8.61e-03	3 1.24e-03
Panel components	Perchloric acid	primary	7.64e-03	3 1.10e-03
Monitor/module	1-methyl-2-pyrrolidinone	primary	7.37e-03	3 1.06e-03
LCD glass mfg.	Hydrofluoric acid	primary	7.33e-03	3 1.05e-03
Monitor/module	Acetic acid	primary	6.79e-03	9.76e-04
Monitor/module	Propylene glycol	primary	6.26e-03	9.00e-04
Panel components	Hydrochloric acid	primary	5.90e-03	8.47e-04
PWB Mfg.	Glycol ethers	model/secondary	5.89e-03	8.46e-04
LCD glass mfg.	Zircon Sand	primary	5.03e-03	3 7.23e-04
Panel components	Toluene	primary	4.60e-03	6.61e-04
Panel components	Heptane	primary	4.33e-03	6.22e-04
Monitor/module	2-ethoxyl ethylacetate	primary	3.55e-03	3 5.10e-04
LCD glass mfg.	Aluminum Oxide	primary	3.12e-02	3 4.49e-04
Monitor/module	Nitrous oxide	primary	2.71e-02	3.90e-04
Monitor/module	Perfluoromethane	primary	2.57e-0	3 70e-04
Monitor/module	Monosilane	primary	2.25e-0	3 23e-04
Panel components	Orthoboric acid	primary	2.12e-0	3 05e-04
Panel components	Hydrochloric acid	primary	2.12e 0.	2 97e-04
Jananese Electric Grid	Uranium vellowcake	model/secondary	2.07e 0	2.97e-04
Monitor/module	Cresol formaldebyde resin	niodel/secondary	1.66e.0	2.940-04
Panel components	A cetope	primary	1.22e.03	2.380-04
Monitor/module	Indium tin ovide	primary	1.220-0.	1.730-04
Womtor/module	Polyothylono mono(nonylphonyl) other	primary	1.030-0.	5 1.516-04
Monitor/module	glycol	primary	9.39e-04	1.35e-04
Panel components	Fuel oil #2	primary	8.15e-04	1.17e-04
Panel components	3 4-difluorobromobenzene	primary	7 31e-04	4 1.05e-04
Panel components	4-pentylphenol	primary	6.84e-04	4 9.83e-05
Panel components	4-bromonhenol	primary	6.53e-04	4 9 38e-05
Monitor/module	I CD material (confidential)	primary	6.21e-04	1 8.93e-05
Monitor/module	1 4-butanolide	primary	5.31e-04	1 7.63e-05
Panel components	3.4.5-trifluorobromobenzene	primary	5.29e-04	7.60e-05
Monitor/module	Hevamethyldisilizane	primary	5.16e-04	1 7.42e-05
Panel components	4.4(propulsyclobevyl)cyclobevapope	primary	1.35e 0/	1 6 26e 05
Panel components	4 propionylphenol	primary	3.800.0	1 5 59e 05
Monitor/module	Hydrogen perovide	primary	2.020.0	1 4.22° 05
Panal components		primary	2.730-0-	4.220-05
Panel components	HCFC 225ch	primary	2.746-04	1 2.020.05
Panel components	Bomoy	primary	1.822.04	1 2.622.05
Panel components	Cyalabayana	primary	1.030-04	2.020-05
Panel components	4 othylphonol	primary	1./8e-04	+ 2.306-03
Monitor/module	4-curyipnenoi Vulana (miyad igan	primary	1.40e-04	+ 2.01e-05
	Ayrene (mixed isomers)	primary	1.04e-04	+ 1.50e-05
Backlight	Dietnyl etner	primary	9.50e-03	1.3/e-05
Panel components	Methyl ethyl ketone	primary	4.61e-03	6.62e-06

Table M-32. LCD LCIA Results for the Chronic Occupational Health Effects Impact Category

			Chronic Occupational	
Process Group	Materials	LCI Data Type	Toxicity (tox-kg)	% of Total
Monitor/module	Triallyl isocyanurate	primary	3.09e-05	4.44e-06
Monitor/module	2,2,4-trimethylpentane	primary	3.05e-05	4.38e-06
US electric grid	Uranium, yellowcake	model/secondary	1.88e-05	2.70e-06
Monitor/module	Ammonium hydroxide	primary	1.03e-05	1.48e-06
Backlight	LNG	primary	8.33e-06	1.20e-06
Monitor/module	2-(2-butoxyethoxy)-ethanol acetate	primary	8.18e-06	1.18e-06
Monitor/module	Sodium dihydrogen phosphate dihydrate	primary	8.12e-06	1.17e-06
Backlight	Mercury	primary	3.99e-06	5.73e-07
Monitor/module	Fluorocarbon resin	primary	3.38e-06	4.86e-07
Monitor/module	Cyclohexane	primary	9.30e-07	1.34e-07
Monitor/module	Methyl ethyl ketone	primary	7.00e-07	1.01e-07
Monitor/module	Aluminum (elemental)	primary	1.70e-11	2.45e-12
Total Manufacturing			6.84e+02	9.83e+01
Use, Maintenance and Repair Life-c	cycle Stage			
US electric grid	Natural gas	model/secondary	1.04e+01	1.50e+00
US electric grid	Petroleum (in ground)	model/secondary	2.84e+00	4.08e-01
US electric grid	Uranium, yellowcake	model/secondary	3.62e-03	5.20e-04
Total Use, Maintenance and Repair			1.33e+01	1.91e+00
End-of-life Life-cycle Stage				
LCD landfilling	Fuel oil #4	primary	9.29e-03	1.34e-03
LCD recycling	Liquified petroleum gas ("propane")	primary	2.76e-03	3.97e-04
US electric grid	Uranium, yellowcake	model/secondary	6.87e-07	9.87e-08
LCD incineration	Fuel oil #4	secondary	-1.95e+00	-2.80e-01
Total End-of-life			-1.94e+00	-2.79e-01
Total All Life-cycle Stages			6.96e+02	1.00e+02

Table M-32. LCD LCIA Results for the Chronic Occupational Health Effects Impact Category

	M-4	LCI Data Tama	Charada Daublia	<u>80- j</u>
Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Materials Processing Life-cycle	Stage			
Invar	Sulfur dioxide	secondary	1.65e+02	8.33e+00
Steel Prod., cold-rolled, semi-finished	Sulfur dioxide	secondary	2.54e+01	1.28e+00
Aluminum Prod.	Sulfur dioxide	secondary	1.38e+01	6.97e-01
Polycarbonate Production	Sulfur dioxide	secondary	7.93e+00	4.01e-01
Lead	Sulfur dioxide	secondary	4.57e+00	2.31e-01
ABS Production	Sulfur dioxide	secondary	2.80e+00	1.41e-01
Ferrite mfg.	Sulfur dioxide	secondary	2.35e+00	1.19e-01
Lead	Arsenic	secondary	2.21e+00	1.11e-01
Polystyrene Prod., high-impact	Sulfur dioxide	secondary	1.20e+00	6.05e-02
Aluminum Prod.	Titanium tetrachloride	secondary	7.20e-01	3.64e-02
Steel Prod., cold-rolled, semi-finished	Carbon monoxide	secondary	3.00e-01	1.51e-02
Invar	Titanium tetrachloride	secondary	2.56e-01	1.29e-02
Lead	Titanium tetrachloride	secondary	2.10e-01	1.06e-02
Steel Prod., cold-rolled, semi-finished	PM	secondary	2.10e-01	1.06e-02
Aluminum Prod.	Manganese cmpds	secondary	1.62e-01	8.18e-03
Invar	Manganese cmpds	secondary	1.50e-01	7.56e-03
Aluminum Prod.	Vanadium	secondary	1.07e-01	5.43e-03
Ferrite mfg.	Manganese cmpds	secondary	9.39e-02	4.74e-03
Invar	Carbon monoxide	secondary	7.60e-02	3.84e-03
Invar	Vanadium	secondary	7.40e-02	3.74e-03
Ferrite mfg.	Carbon monoxide	secondary	6.32e-02	3.19e-03
Steel Prod., cold-rolled, semi-finished	Vanadium	secondary	5.15e-02	2.60e-03
Aluminum Prod.	Arsenic cmpds	secondary	4.75e-02	2.40e-03
Lead	Manganese cmpds	secondary	4.60e-02	2.33e-03
Polycarbonate Production	Carbon monoxide	secondary	4.48e-02	2.27e-03
Polycarbonate Production	Methane	secondary	4.06e-02	2.05e-03
Polycarbonate Production	Nitrogen dioxide	secondary	3.88e-02	1.96e-03
Invar	Arsenic	secondary	3.21e-02	1.62e-03
Ferrite mfg.	Arsenic	secondary	2.90e-02	1.46e-03
Steel Prod., cold-rolled, semi-finished	Nitrogen dioxide	secondary	2.80e-02	1.41e-03
Lead	Zinc (elemental)	secondary	2.42e-02	1.22e-03
Ferrite mfg.	Titanium tetrachloride	secondary	2.36e-02	1.19e-03
Invar	Zinc (elemental)	secondary	2.28e-02	1.15e-03
Ferrite mfg.	Zinc (elemental)	secondary	2.21e-02	1.12e-03
Steel Prod., cold-rolled, semi-finished	Methane	secondary	2.18e-02	1.10e-03
ABS Production	Carbon monoxide	secondary	2.17e-02	1.10e-03
Lead	Vanadium	secondary	2.09e-02	1.06e-03
Lead	Carbon monoxide	secondary	2.04e-02	1.03e-03
Steel Prod., cold-rolled, semi-finished	Fluorides (F-)	secondary	1.92e-02	9.70e-04
Aluminum Prod.	Carbon monoxide	secondary	1.86e-02	9.41e-04
Aluminum Prod.	PM	secondary	1.78e-02	8.99e-04
Styrene-butadiene Copolymer Prod.	Carbon monoxide	secondary	1.73e-02	8.74e-04
Invar	Arsenic cmpds	secondary	1.69e-02	8.53e-04
Aluminum Prod.	Methane	secondary	1.61e-02	8.13e-04
Steel Prod., cold-rolled, semi-finished	Manganese cmpds	secondary	1.49e-02	7.53e-04
Ferrite mfg.	Vanadium	secondary	1.45e-02	7.33e-04
Aluminum Prod.	Nitrogen dioxide	secondary	1.43e-02	7.21e-04
Styrene-butadiene Copolymer Prod.	Nitrogen oxides	secondary	1.40e-02	7.07e-04
Lead	Arsenic cmpds	secondary	1.38e-02	6.98e-04

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Styrene-butadiene Copolymer Prod.	Methane	secondary	1.37e-02	6.94e-04
Polycarbonate Production	PM	secondary	1.29e-02	6.53e-04
Invar	Methane	secondary	1.15e-02	5.82e-04
Styrene-butadiene Copolymer Prod.	Sulfur oxides	secondary	1.14e-02	5.77e-04
Aluminum Prod.	Barium cmpds	secondary	1.04e-02	5.28e-04
ABS Production	Methane	secondary	1.02e-02	5.14e-04
Invar	Nitrogen dioxide	secondary	1.01e-02	5.12e-04
ABS Production	Nitrogen dioxide	secondary	9.32e-03	4.71e-04
Lead	Nitrogen dioxide	secondary	8.37e-03	4.23e-04
Aluminum Prod.	Selenium	secondary	6.29e-03	3.18e-04
Ferrite mfg.	Methane	secondary	5.71e-03	2.88e-04
Invar	PM	secondary	5.51e-03	2.78e-04
Lead	Methane	secondary	5.04e-03	2.55e-04
Steel Prod., cold-rolled, semi-finished	Arsenic	secondary	4.52e-03	2.28e-04
Invar	Barium cmpds	secondary	4.31e-03	2.18e-04
Steel Prod., cold-rolled, semi-finished	Phosphorus (yellow or white)	secondary	4.07e-03	2.06e-04
Ferrite mfg.	Nitrogen dioxide	secondary	4.05e-03	2.05e-04
Polystyrene Prod., high-impact	Nitrogen dioxide	secondary	3.63e-03	1.83e-04
Ferrite mfg.	PM	secondary	3.44e-03	1.74e-04
Polystyrene Prod., high-impact	Methane	secondary	3.33e-03	1.68e-04
Aluminum Prod.	Aluminum (+3)	secondary	3.18e-03	1.60e-04
Lead	Lead	secondary	3.16e-03	1.60e-04
ABS Production	Hydrochloric acid	secondary	2.93e-03	1.48e-04
Lead	Barium cmpds	secondary	2.71e-03	1.37e-04
ABS Production	PM	secondary	2.54e-03	1.28e-04
Polystyrene Prod high-impact	Carbon monoxide	secondary	2.45e-03	1.24e-04
Invar	Selenium	secondary	2.32e-03	1.17e-04
Steel Prod cold-rolled semi-finished	Hydrochloric acid	secondary	2.05e-03	1.03e-04
Styrene-butadiene Copolymer Prod	PM	secondary	1 99e-03	1 00e-04
Invar	Hydrochloric acid	secondary	1.85e-03	9.35e-05
Lead	Selenium	secondary	1 84e-03	9.27e-05
Aluminum Prod	Hydrochloric acid	secondary	1 78e-03	9.02e-05
Steel Prod cold-rolled semi-finished	Benzene	secondary	1.76e-03	8 89e-05
Steel Prod. cold-rolled semi-finished	Barium cmpds	secondary	1.73e-03	8 75e-05
Lead	PM	secondary	1.75e 03	8 50e-05
Ferrite mfg	Arsenic cmpds	secondary	1.64e-03	8 28e-05
ABS Production	Formaldehyde	secondary	1.60e-03	8.09e-05
Steel Prod. cold-rolled semi-finished	Nitrous oxide	secondary	1.55e-03	7.85e-05
Steel Prod. cold-rolled semi-finished	Arsenic cmpds	secondary	1.55e 03	7.85e 05
Steel Prod. cold-rolled semi-finished	Ammonia	secondary	1.5 fe 03	6 38e-05
Aluminum Prod	Cadmium cmpds	secondary	1.20e 03	5.92e-05
Invor	Aluminum (+3)	secondary	1.170 03	5.72e 05
Lead	Hydrochloric acid	secondary	1.14c-03	5.41e-05
Steel Prod cold-rolled semi-finished	Silicon	secondary	1.07e-03	5.12e-05
Invar	Copper	secondary	0.80-04	A 00e 05
Lead	Aluminum (+3)	secondary	9.090-04	4.570-05
Invar	Benzene	secondary	0.180.04	4.640.05
Steel Prod cold rolled comi finished	Titanium tatrachlarida	secondary	9.100-04	4.210.05
Ferrite mfg	Barium empds	secondary	0.35e-04	4.210-03
ronne illig.	Bartum empus	secondary	/.010-04	3.936-03

Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Ferrite mfg.	Hydrochloric acid	secondary	6.93e-04	3.50e-05
Polystyrene Prod., high-impact	PM	secondary	6.05e-04	3.05e-05
Ferrite mfg.	Benzene	secondary	5.86e-04	2.96e-05
Aluminum Prod.	Benzene	secondary	5.70e-04	2.88e-05
Steel Prod., cold-rolled, semi-finished	Ethane	secondary	4.92e-04	2.48e-05
ABS Production	Hydrofluoric acid	secondary	4.87e-04	2.46e-05
Aluminum Prod.	Hydrofluoric acid	secondary	4.81e-04	2.43e-05
Polycarbonate Production	Hydrochloric acid	secondary	4.65e-04	2.35e-05
Aluminum Prod.	Barium sulfate	secondary	4.54e-04	2.29e-05
Invar	Silicon	secondary	4.52e-04	2.28e-05
Steel Prod., cold-rolled, semi-finished	Formaldehyde	secondary	4.15e-04	2.10e-05
ABS Production	Aromatic hydrocarbons	secondary	3.81e-04	1.93e-05
Steel Prod., cold-rolled, semi-finished	Titanium	secondary	3.30e-04	1.67e-05
Polycarbonate Production	Sulfuric acid	secondary	3.18e-04	1.60e-05
ABS Production	Ammonia	secondary	3.07e-04	1.55e-05
Steel Prod., cold-rolled, semi-finished	Selenium	secondary	3.00e-04	1.51e-05
Lead	Benzene	secondary	2.99e-04	1.51e-05
Ferrite mfg.	Silicon	secondary	2.97e-04	1.50e-05
Invar	Phosphorus (yellow or white)	secondary	2.89e-04	1.46e-05
Ferrite mfg.	Selenium	secondary	2.86e-04	1.44e-05
Ferrite mfg.	Phosphorus (yellow or white)	secondary	2.85e-04	1.44e-05
Aluminum Prod.	Nitrous oxide	secondary	2.76e-04	1.40e-05
Aluminum Prod.	Perfluoromethane	secondary	2.59e-04	1.31e-05
Polycarbonate Production	Mercury compounds	secondary	2.44e-04	1.23e-05
ABS Production	Ethane	secondary	2.37e-04	1.20e-05
Aluminum Prod.	Titanium	secondary	2.30e-04	1.16e-05
Aluminum Prod.	Zinc (elemental)	secondary	2.30e-04	1.16e-05
Aluminum Prod.	Silicon	secondary	2.19e-04	1.10e-05
Steel Prod., cold-rolled, semi-finished	Zinc (elemental)	secondary	2.16e-04	1.09e-05
Invar	Titanium	secondary	2.16e-04	1.09e-05
Steel Prod., cold-rolled, semi-finished	Tin (Sn++, Sn4+)	secondary	2.09e-04	1.05e-05
Steel Prod., cold-rolled, semi-finished	Molybdenum	secondary	1.88e-04	9.50e-06
Aluminum Prod.	Copper (+1 & +2)	secondary	1.80e-04	9.10e-06
Aluminum Prod.	Strontium (Sr II)	secondary	1.70e-04	8.57e-06
Invar	Ethane	secondary	1.65e-04	8.35e-06
Polycarbonate Production	Aromatic hydrocarbons	secondary	1.64e-04	8.30e-06
Ferrite mfg.	Ethane	secondary	1.61e-04	8.15e-06
Invar	Formaldehyde	secondary	1.60e-04	8.09e-06
Ferrite mfg.	Formaldehyde	secondary	1.56e-04	7.90e-06
Invar	Molvbdenum	secondary	1.52e-04	7.69e-06
ABS Production	Sulfuric acid	secondary	1.46e-04	7.36e-06
Lead	Silicon	secondary	1.43e-04	7.20e-06
Styrene-butadiene Copolymer Prod.	Sulfuric acid	secondary	1.42e-04	7.19e-06
Invar	Nitrates/nitrites	secondary	1.37e-04	6.91e-06
Styrene-butadiene Copolymer Prod.	Aromatic hydrocarbons	secondary	1.36e-04	6.86e-06
Invar	Cadmium cmpds	secondary	1.27e-04	6.42e-06
Steel Prod., cold-rolled, semi-finished	Barium	secondary	1 23e-04	6.22e-06
Lead	Barium sulfate	secondary	1 15e-04	5.79e-06
ABS Production	Mercury compounds	secondary	1.12e-04	5.64e-06
Ferrite mfg.	Aluminum (+3)	secondary	1 12e-04	5.63e-06
B.		secondary	1.120 04	2.050 00

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Styrene-butadiene Copolymer Prod.	Mercury compounds	secondary	1.09e-04	5.51e-06
ABS Production	Aluminum (+3)	secondary	1.02e-04	5.14e-06
Steel Prod., cold-rolled, semi-finished	Antimony	secondary	9.54e-05	4.82e-06
Invar	Copper (+1 & +2)	secondary	9.43e-05	4.77e-06
Invar	Nickel	secondary	9.29e-05	4.69e-06
Ferrite mfg.	Titanium	secondary	9.23e-05	4.66e-06
Polycarbonate Production	Fluorides (F-)	secondary	9.20e-05	4.65e-06
Aluminum Prod.	Nitrate	secondary	9.13e-05	4.61e-06
Invar	Strontium (Sr II)	secondary	8.79e-05	4.44e-06
Steel Prod., cold-rolled, semi-finished	Boron	secondary	8.57e-05	4.33e-06
Steel Prod., cold-rolled, semi-finished	Strontium (Sr II)	secondary	7.92e-05	4.00e-06
Invar	Lead	secondary	7.24e-05	3.66e-06
Ferrite mfg.	Lead	secondary	6.92e-05	3.50e-06
Styrene-butadiene Copolymer Prod.	Hydrochloric acid	secondary	6.82e-05	3.45e-06
Invar	Barium	secondary	6.53e-05	3.30e-06
Lead	Ethane	secondary	6.32e-05	3.19e-06
Steel Prod., cold-rolled, semi-finished	Silver compounds	secondary	6.17e-05	3.12e-06
Invar	Tin $(Sn++, Sn4+)$	secondary	6.07e-05	3.07e-06
Polystyrene Prod., high-impact	Aromatic hydrocarbons	secondary	6.05e-05	3.05e-06
ABS Production	Nitrate	secondary	6.01e-05	3.04e-06
Steel Prod., cold-rolled, semi-finished	Nitrates/nitrites	secondary	5.66e-05	2.86e-06
Invar	Hydrofluoric acid	secondary	5 37e-05	2.72e-06
Ferrite mfg	Molyhdenum	secondary	5.24e-05	2.65e-06
Lead	Conner $(+1 \& +2)$	secondary	5.21e-05	2.63e-06
Steel Prod cold-rolled semi-finished	Chromium (VI)	secondary	5.17e-05	2.61e-06
Invar	Ammonia	secondary	5.01e-05	2.53e-06
Invar	$Z_{inc}(+2)$	secondary	4 87e-05	2.55 e 00
Invar	Boron	secondary	4.87e-05	2.10e 00
Steel Prod cold-rolled semi-finished	Acetylene	secondary	4.66e-05	2.15e 00
Steel Prod. cold-rolled semi-finished	Mercury compounds	secondary	4.53e-05	2.35e 00
Lead	Cadmium	secondary	4 52e-05	2.29e-06
Styrene-butadiene Conolymer Prod	Fluorides (F-)	secondary	4.12e-05	2.29e-06
Lead	Nitrous oxide	secondary	4.12e 05	2.05e-06
Aluminum Prod	Barium	secondary	3.91e-05	1.98e-06
Steel Prod cold-rolled semi-finished	Hydrofluoric acid	secondary	3.79e-05	1.91e-06
Ferrite mfg	Barium	secondary	3.73e-05	1.89e-06
Ferrite mfg	Zinc(+2)	secondary	3.75e-05	1.85e-06
Aluminum Prod	Zinc(+2)	secondary	3.48e.05	1.76e.06
Invor	Barium sulfate	secondary	3.460.05	1.75e.06
Staal Brad, and ralled somi finished	Barium sulfate	secondary	3.400-05	1.730.06
L and	Strontium (Sr II)	secondary	3.43C-03	1.730-00
Lead	Darium gulfata	secondary	3.400-03	1.72e-00
Ferrite mig.	A low income (12)	secondary	3.388-03	1./1e-06
Lood	Anuminum (+3)	secondary	5.15e-05	1.596-06
	Barium	secondary	2.886-05	1.40e-06
Aluminum Prod.	Aromatic hydrocarbons	secondary	2.83e-05	1.43e-06
Lead	Copper	secondary	2.82e-05	1.42e-06
	Hydroffuoric acid	secondary	2.73e-05	1.38e-06
Steel Prod., cold-rolled, semi-finished	Hydrogen sulfide	secondary	2.72e-05	1.3/e-06
Aluminum Prod.	Nickel cmpds	secondary	2.71e-05	1.37e-06
Invar	Antimony	secondary	2.70e-05	1.37e-06

Process Group	Material	LCI Data Type	Chronic Public	% of Total
Steel Prod. cold-rolled semi-finished	Cadmium cmpds	secondary	2 70e-05	1 36e-06
Ferrite mfg	Antimony	secondary	2.70e 05	1.33e-06
Ferrite mfg	Boron	secondary	2.010 05 2.58e-05	1.30e-06
Ferrite mfg	Strontium (Sr II)	secondary	2.50e-05	1.30e-06
Aluminum Prod	Nickel	secondary	2.57e-05	1.30e 00
Steel Prod cold-rolled semi-finished	Aluminum (elemental)	secondary	2.52e 05	1.25e-06
Aluminum Prod.	Ammonia	secondary	2.45e-05	1.24e-06
Polystyrene Prod., high-impact	Hydrochloric acid	secondary	2.42e-05	1.22e-06
Aluminum Prod.	Copper	secondary	2.38e-05	1.20e-06
Invar	Nickel cmpds	secondary	2.15e-05	1.09e-06
Ferrite mfg.	Nitrous oxide	secondary	2.12e-05	1.07e-06
ABS Production	Hydrogen sulfide	secondary	2.02e-05	1.02e-06
Lead	Ammonia	secondary	1.88e-05	9.49e-07
Invar	Silver compounds	secondary	1.86e-05	9.42e-07
Ferrite mfg.	Silver compounds	secondary	1.82e-05	9.20e-07
Aluminum Prod.	Fluoride	secondary	1.82e-05	9.19e-07
Lead	Nitrate	secondary	1.80e-05	9.10e-07
Aluminum Prod.	Lead cmpds	secondary	1.74e-05	8.81e-07
Steel Prod., cold-rolled, semi-finished	Pentane	secondary	1.70e-05	8.57e-07
Invar	Aromatic hydrocarbons	secondary	1.67e-05	8.44e-07
Lead	Boron	secondary	1.66e-05	8.37e-07
Aluminum Prod.	Vanadium (V3+, V5+)	secondary	1.63e-05	8.23e-07
Ferrite mfg.	Ammonia	secondary	1.62e-05	8.20e-07
Invar	Aluminum (elemental)	secondary	1.52e-05	7.70e-07
Aluminum Prod.	Nitrites	secondary	1.47e-05	7.40e-07
Invar	Hydrogen sulfide	secondary	1.45e-05	7.34e-07
Invar	Chromium (VI)	secondary	1.45e-05	7.31e-07
Aluminum Prod.	Perfluoroethane	secondary	1.44e-05	7.28e-07
Steel Prod., cold-rolled, semi-finished	Aromatic hydrocarbons	secondary	1.43e-05	7.24e-07
Aluminum Prod.	Chromium (VI)	secondary	1.39e-05	7.03e-07
Invar	Cadmium	secondary	1.34e-05	6.77e-07
Invar	Acetylene	secondary	1.32e-05	6.68e-07
Ferrite mfg.	Acetylene	secondary	1.29e-05	6.52e-07
Ferrite mfg.	Hydrofluoric acid	secondary	1.27e-05	6.44e-07
Steel Prod., cold-rolled, semi-finished	Copper	secondary	1.22e-05	6.17e-07
Steel Prod., cold-rolled, semi-finished	Nickel	secondary	1.16e-05	5.84e-07
Ferrite mfg.	Cadmium	secondary	1.11e-05	5.61e-07
Aluminum Prod.	Aluminum (elemental)	secondary	1.05e-05	5.32e-07
Polycarbonate Production	Copper (+1 & +2)	secondary	1.04e-05	5.24e-07
Ferrite mfg.	Hydrogen sulfide	secondary	1.03e-05	5.22e-07
Lead	Zinc (+2)	secondary	9.93e-06	5.02e-07
Aluminum Prod.	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI) secondary	9.88e-06	4.99e-07
Polycarbonate Production	Phenol	secondary	9.34e-06	4.72e-07
Invar	Pentane	secondary	8.97e-06	4.53e-07
Lead	Fluoride	secondary	8.91e-06	4.50e-07
Steel Prod., cold-rolled, semi-finished	Hexane	secondary	8.86e-06	4.48e-07
Invar	Fluoride	secondary	8.80e-06	4.45e-07
Ferrite mfg.	Pentane	secondary	8.76e-06	4.43e-07
Steel Prod., cold-rolled, semi-finished	Copper (+1 & +2)	secondary	8.59e-06	4.34e-07
Lead	Aluminum (elemental)	secondary	8.16e-06	4.12e-07

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Ferrite mfg.	Nitrate	secondary	8.08e-06	4.08e-07
Lead	Nickel cmpds	secondary	7.95e-06	4.01e-07
Ferrite mfg.	Aluminum (elemental)	secondary	7.77e-06	3.92e-07
Invar	Benzo[a]pyrene	secondary	7.72e-06	3.90e-07
Aluminum Prod.	Hydrogen sulfide	secondary	7.53e-06	3.80e-07
Polycarbonate Production	Hydrofluoric acid	secondary	7.39e-06	3.73e-07
Aluminum Prod.	Acetic acid	secondary	7.33e-06	3.71e-07
Lead	Nickel	secondary	6.71e-06	3.39e-07
Ferrite mfg.	Chromium (VI)	secondary	6.52e-06	3.30e-07
Invar	Nitrites	secondary	6.45e-06	3.26e-07
Invar	Acetic acid	secondary	6.08e-06	3.07e-07
Invar	Vanadium (V3+, V5+)	secondary	6.00e-06	3.03e-07
Invar	Lead cmpds	secondary	5.96e-06	3.01e-07
Ferrite mfg.	Copper (+1 & +2)	secondary	5.96e-06	3.01e-07
Ferrite mfg.	Cadmium cmpds	secondary	5.94e-06	3.00e-07
Lead	Hydrogen sulfide	secondary	5.93e-06	3.00e-07
Polycarbonate Production	Ammonia	secondary	5.86e-06	2.96e-07
Lead	Chromium (VI)	secondary	5.80e-06	2.93e-07
Styrene-butadiene Copolymer Prod.	Nitrate	secondary	5.79e-06	2.93e-07
Lead	Lead cmpds	secondary	5.50e-06	2.78e-07
ABS Production	Heptane	secondary	5.02e-06	2.53e-07
ABS Production	Copper $(+1 \& +2)$	secondary	4.75e-06	2.40e-07
Lead	Vanadium (V3+, V5+)	secondary	4.75e-06	2.40e-07
Styrene-butadiene Copolymer Prod.	Copper (+1 & +2)	secondary	4.64e-06	2.35e-07
Ferrite mfg.	Copper	secondary	4.55e-06	2.30e-07
Ferrite mfg.	Aromatic hydrocarbons	secondary	4.52e-06	2.29e-07
Steel Prod., cold-rolled, semi-finished	Bromine	secondary	4.06e-06	2.05e-07
Polycarbonate Production	Halogenated hydrocarbons (unspecified)	secondary	3.69e-06	1.87e-07
Polycarbonate Production	Hydrogen sulfide	secondary	3.69e-06	1.87e-07
Steel Prod., cold-rolled, semi-finished	Acetic acid	secondary	3.56e-06	1.80e-07
Invar	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI)	secondary	3.45e-06	1.74e-07
Aluminum Prod.	Cadmium	secondary	3.36e-06	1.70e-07
Steel Prod., cold-rolled, semi-finished	Boric acid	secondary	3.30e-06	1.67e-07
Ferrite mfg.	Nickel	secondary	3.26e-06	1.65e-07
Lead	Nitrites	secondary	3 10e-06	1 57e-07
Steel Prod., cold-rolled, semi-finished	Strontium	secondary	3.01e-06	1.52e-07
Lead	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI)	secondary	2.99e-06	1.51e-07
Invar	Cobalt (Co I Co II Co III)	secondary	2.88e-06	1.46e-07
Aluminum Prod	Triethylene glycol	secondary	2.60e-06	1.16e-07
Invar	Hexane	secondary	2.58e-06	1.30e-07
Polystyrene Prod high-impact	Ammonia	secondary	2.56e-06	1.30e 07
Ferrite mfg	Hevane	secondary	2.50e 00	1.27e-07
Aluminum Prod	Lead	secondary	2.51e 00	1.2/e 07
Steel Prod cold-rolled semi-finished	Lead	secondary	2.10e-06	1.11e-07
Inver	Boron (B III)	secondary	2.20e-00 2.18e-06	1.10e-07
Lead	Bromine	secondary	2.166-00	1.00-07
Invar	Triethylene glycol	secondary	2.100-00	1.090-07
Steel Prod cold-rolled semi finished	Triethylene glycol	secondary	2.140-00	1.030-07
Ferrite mfg	Triethylene glycol	secondary	2.120-00	1.070-07
I and	Triethylene glycol	secondary	2.090-00	1.010.07
Luau	richylene giyeoi	secondary	2.018-00	1.01e-07

 Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Steel Prod., cold-rolled, semi-finished	Cadmium	secondary	1.98e-06	1.00e-07
Invar	Methanol	secondary	1.87e-06	9.43e-08
Polycarbonate Production	Chlorine	secondary	1.85e-06	9.33e-08
Ferrite mfg.	Fluoride	secondary	1.85e-06	9.33e-08
Lead	Cobalt (Co I, Co II, Co III)	secondary	1.84e-06	9.28e-08
Aluminum Prod.	Toluene	secondary	1.81e-06	9.15e-08
Steel Prod., cold-rolled, semi-finished	Lead cmpds	secondary	1.77e-06	8.92e-08
Steel Prod., cold-rolled, semi-finished	Uranium	secondary	1.71e-06	8.65e-08
Styrene-butadiene Copolymer Prod.	Aluminum (+3)	secondary	1.65e-06	8.36e-08
Invar	Strontium	secondary	1.59e-06	8.04e-08
Ferrite mfg.	Acetic acid	secondary	1.52e-06	7.67e-08
Steel Prod., cold-rolled, semi-finished	Toluene	secondary	1.52e-06	7.67e-08
Lead	Mercury	secondary	1.51e-06	7.64e-08
Invar	Bromium (Br)	secondary	1.42e-06	7.16e-08
Ferrite mfg.	Bromine	secondary	1.38e-06	6.99e-08
Aluminum Prod.	Xylene (mixed isomers)	secondary	1.38e-06	6.96e-08
Lead	Boron (B III)	secondary	1.31e-06	6.63e-08
ABS Production	Chlorine	secondary	1.27e-06	6.42e-08
Lead	Acetic acid	secondary	1 27e-06	6 41e-08
Steel Prod cold-rolled semi-finished	Fthylene	secondary	1.27e-06	6 20e-08
Steel Prod. cold-rolled semi-finished	Cvanide (-1)	secondary	1.250 00	6 10e-08
Steel Prod. cold-rolled semi-finished	Methanol	secondary	1 19e-06	6.03e-08
Ferrite mfg	Nickel cmpds	secondary	1.13e-06	5.72e-08
Steel Prod cold-rolled semi-finished	Xylene (mixed isomers)	secondary	1.13e-06	5.67e-08
Steel Prod. cold-rolled semi-finished	Naphthalene	secondary	1.12e-06	5.67e-08
Invar	Toluene	secondary	1.11e-06	5.55e-08
Steel Prod cold-rolled semi-finished	Zinc (+2)	secondary	1.100 00	5.35e-08
Ferrite mfg	Lead cmpds	secondary	1.000-00	5.25e-08
Steel Prod cold rolled semi finished	Phenol	secondary	9.70e.07	4 90e 08
Invar	Boric acid	secondary	9.35e-07	4.72e-08
Steel Prod cold rolled semi finished	Hentane	secondary	9.330-07	4.720-08
Palvaerbaneta Production	A luminum (+2)	secondary	9.330-07	4.720-08
Polycarbonate Production	Ethonethiol	secondary	9.230-07	4.67e.08
Polycarbonate Production	Land	secondary	9.236-07	4.670.08
Polycarbonate Production	Nitrata	secondary	9.236-07	4.070-08
Polycarbonate Production	Nitrous ovide	secondary	9.236-07	4.672.08
Polycarbonate Production	Zing (12)	secondary	9.236-07	4.07e-08
	Zinc (+2)	secondary	9.23e-07	4.676-08
Ferrite mig.	Boric acid	secondary	9.13e-07	4.61e-08
	Manganese	secondary	8.4/e-0/	4.28e-08
Steel Prod., cold-rolled, semi-finished	Nickel cmpds	secondary	8.46e-07	4.2/e-08
Ferrite mfg.	Strontium	secondary	8.38e-07	4.23e-08
Styrene-butadiene Copolymer Prod.	Chlorine	secondary	8.27e-07	4.18e-08
Styrene-butadiene Copolymer Prod.	Hydrofluoric acid	secondary	8.27e-07	4.18e-08
Aluminum Prod.	Cobalt	secondary	8.23e-07	4.16e-08
Steel Prod., cold-rolled, semi-finished	Cobalt	secondary	8.10e-07	4.09e-08
Lead	Mercury compounds	secondary	7.89e-07	3.99e-08
Invar	Xylene (mixed isomers)	secondary	7.67e-07	3.88e-08
Aluminum Prod.	Strontium	secondary	7.60e-07	3.84e-08
Steel Prod., cold-rolled, semi-finished	Beryllium	secondary	7.21e-07	3.64e-08
ABS Production	Nickel cmpds	secondary	7.16e-07	3.62e-08

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Ferrite mfg.	Vanadium (V3+, V5+)	secondary	6.83e-07	3.45e-08
Aluminum Prod.	Phenol	secondary	6.39e-07	3.23e-08
Invar	Manganese	secondary	6.32e-07	3.19e-08
Polystyrene Prod., high-impact	Nitrate	secondary	6.05e-07	3.05e-08
Invar	Ethylene	secondary	6.01e-07	3.03e-08
Steel Prod., cold-rolled, semi-finished	Vanadium (V3+, V5+)	secondary	5.94e-07	3.00e-08
ABS Production	Phenol	secondary	5.88e-07	2.97e-08
Ferrite mfg.	Ethylene	secondary	5.79e-07	2.92e-08
Styrene-butadiene Copolymer Prod.	Ammonia	secondary	5.62e-07	2.84e-08
Steel Prod., cold-rolled, semi-finished	Benzo[a]pyrene	secondary	5.60e-07	2.83e-08
Ferrite mfg.	Toluene	secondary	5.31e-07	2.68e-08
ABS Production	HALON-1301	secondary	5.12e-07	2.59e-08
Styrene-butadiene Copolymer Prod.	Phenol	secondary	4.92e-07	2.49e-08
Invar	Uranium	secondary	4.89e-07	2.47e-08
Ferrite mfg.	Uranium	secondary	4.77e-07	2.41e-08
Steel Prod., cold-rolled, semi-finished	Boron (B III)	secondary	4.70e-07	2.37e-08
Polycarbonate Production	Mercury	secondary	4.62e-07	2.33e-08
Ferrite mfg.	Nitrites	secondary	4.48e-07	2.26e-08
Invar	Phenol	secondary	4.41e-07	2.23e-08
ABS Production	Ethanethiol	secondary	4.24e-07	2.14e-08
ABS Production	Fluoride	secondary	4.24e-07	2.14e-08
ABS Production	Halogenated hydrocarbons (unspecified)	secondary	4.24e-07	2.14e-08
ABS Production	Lead	secondary	4.24e-07	2.14e-08
ABS Production	Nitrous oxide	secondary	4.24e-07	2.14e-08
ABS Production	Zinc (+2)	secondary	4.24e-07	2.14e-08
Styrene-butadiene Copolymer Prod.	Halogenated hydrocarbons (unspecified)	secondary	4.14e-07	2.09e-08
Styrene-butadiene Copolymer Prod.	Hydrogen sulfide	secondary	4.14e-07	2.09e-08
Styrene-butadiene Copolymer Prod.	Lead	secondary	4.14e-07	2.09e-08
Styrene-butadiene Copolymer Prod.	Nitrous oxide	secondary	4.14e-07	2.09e-08
Styrene-butadiene Copolymer Prod.	Zinc (+2)	secondary	4.14e-07	2.09e-08
Aluminum Prod.	HALON-1301	secondary	3.61e-07	1.82e-08
Steel Prod., cold-rolled, semi-finished	Manganese	secondary	3.60e-07	1.82e-08
Ferrite mfg.	Xylene (mixed isomers)	secondary	3.50e-07	1.77e-08
Ferrite mfg.	Methanol	secondary	3.36e-07	1.70e-08
Invar	Cyanide (-1)	secondary	2.89e-07	1.46e-08
Invar	Heptane	secondary	2.83e-07	1.43e-08
Invar	Acetaldehyde	secondary	2.79e-07	1.41e-08
Ferrite mfg.	Heptane	secondary	2.76e-07	1.40e-08
Lead	Xylene (mixed isomers)	secondary	2.54e-07	1.29e-08
Aluminum Prod.	1,2-Dichlorotetrafluoroethane	secondary	2.49e-07	1.26e-08
Steel Prod., cold-rolled, semi-finished	Rubidium ion (Rb+)	secondary	2.42e-07	1.22e-08
Ferrite mfg.	Cobalt	secondary	2.40e-07	1.21e-08
Invar	Mercury	secondary	2.12e-07	1.07e-08
ABS Production	Mercury	secondary	2.12e-07	1.07e-08
Styrene-butadiene Copolymer Prod.	Mercury	secondary	2.07e-07	1.05e-08
Ferrite mfg.	Cyanide (-1)	secondary	2.05e-07	1.04e-08
Invar	Beryllium	secondary	2.05e-07	1.04e-08
Aluminum Prod.	Mercury	secondary	2.02e-07	1.02e-08
Ferrite mfg.	Beryllium	secondary	2.00e-07	1.01e-08
Ferrite mfg.	Phenol	secondary	1.95e-07	9.87e-09
0				

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Aluminum Prod.	Cyanide (-1)	secondary	1.89e-07	9.57e-09
Steel Prod., cold-rolled, semi-finished	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI)	secondary	1.81e-07	9.17e-09
Polystyrene Prod., high-impact	Phenol	secondary	1.80e-07	9.09e-09
Steel Prod., cold-rolled, semi-finished	Morpholine	secondary	1.79e-07	9.05e-09
Steel Prod., cold-rolled, semi-finished	Propylene	secondary	1.77e-07	8.94e-09
Ferrite mfg.	Cobalt (Co I, Co II, Co III)	secondary	1.76e-07	8.88e-09
Lead	Methanol	secondary	1.62e-07	8.21e-09
Steel Prod., cold-rolled, semi-finished	Acetaldehyde	secondary	1.62e-07	8.18e-09
Ferrite mfg.	Mercury	secondary	1.62e-07	8.16e-09
Ferrite mfg.	Benzo[a]pyrene	secondary	1.55e-07	7.85e-09
Ferrite mfg.	Boron (B III)	secondary	1.55e-07	7.83e-09
ABS Production	Ethylene	secondary	1.52e-07	7.68e-09
Invar	HALON-1301	secondary	1.52e-07	7.68e-09
Lead	Cobalt	secondary	1.46e-07	7.39e-09
Lead	Phenol	secondary	1.37e-07	6.94e-09
Invar	Ethylbenzene	secondary	1.37e-07	6.90e-09
Invar	Acetone	secondary	1.36e-07	6.88e-09
Lead	Aromatic hydrocarbons	secondary	1.36e-07	6.86e-09
Aluminum Prod.	Chromium (III)	secondary	1.31e-07	6.62e-09
Ferrite mfg.	Manganese	secondary	1.19e-07	6.00e-09
Steel Prod., cold-rolled, semi-finished	Fluorine	secondary	1.14e-07	5.78e-09
Steel Prod., cold-rolled, semi-finished	Lanthanum	secondary	1.07e-07	5.43e-09
Invar	1,2-Dichlorotetrafluoroethane	secondary	1.03e-07	5.22e-09
Lead	Cyanide (-1)	secondary	9.76e-08	4.93e-09
Invar	Nitrous oxide	secondary	9.70e-08	4.90e-09
Lead	Ethylene	secondary	9.43e-08	4.77e-09
Lead	Toluene	secondary	8.24e-08	4.16e-09
Steel Prod., cold-rolled, semi-finished	Acetone	secondary	7.91e-08	3.99e-09
Invar	Rubidium ion (Rb+)	secondary	7.31e-08	3.69e-09
Ferrite mfg.	Rubidium ion (Rb+)	secondary	7.14e-08	3.61e-09
Lead	HALON-1301	secondary	6.78e-08	3.42e-09
Invar	Sulfuric acid	secondary	5.99e-08	3.03e-09
Ferrite mfg.	Sulfuric acid	secondary	5.85e-08	2.95e-09
Steel Prod., cold-rolled, semi-finished	Thorium	secondary	5.77e-08	2.91e-09
Invar	Mercury compounds	secondary	5.58e-08	2.82e-09
Ferrite mfg.	Mercury compounds	secondary	5.45e-08	2.75e-09
Steel Prod., cold-rolled, semi-finished	Ethylbenzene	secondary	5.25e-08	2.65e-09
Ferrite mfg.	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI)	secondary	5.09e-08	2.57e-09
Invar	Morpholine	secondary	5.07e-08	2.56e-09
Steel Prod., cold-rolled, semi-finished	HALON-1301	secondary	5.04e-08	2.55e-09
Invar	Propylene	secondary	5.03e-08	2.54e-09
Ferrite mfg.	Morpholine	secondary	4.95e-08	2.50e-09
Ferrite mfg.	Propylene	secondary	4.91e-08	2.48e-09
Invar	Chromium (III)	secondary	4.79e-08	2.42e-09
Lead	Ethylbenzene	secondary	4.72e-08	2.39e-09
Ferrite mfg.	Acetaldehyde	secondary	4.63e-08	2.34e-09
Lead	Chromium (III)	secondary	3.86e-08	1.95e-09
Lead	Acetaldehyde	secondary	3.59e-08	1.81e-09
Steel Prod., cold-rolled, semi-finished	Tin	secondary	3.35e-08	1.69e-09
Steel Prod., cold-rolled, semi-finished	Mercury	secondary	3.26e-08	1.65e-09

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Invar	Fluorine	secondary	3.24e-08	1.64e-09
Steel Prod., cold-rolled, semi-finished	Phosphorus pentoxide	secondary	3.22e-08	1.63e-09
Ferrite mfg.	Fluorine	secondary	3.16e-08	1.60e-09
Invar	Lanthanum	secondary	3.06e-08	1.55e-09
Ferrite mfg.	Lanthanum	secondary	2.99e-08	1.51e-09
Steel Prod., cold-rolled, semi-finished	Hydrazine	secondary	2.65e-08	1.34e-09
Steel Prod., cold-rolled, semi-finished	Scandium	secondary	2.61e-08	1.32e-09
Steel Prod., cold-rolled, semi-finished	Chlorine	secondary	2.28e-08	1.15e-09
Ferrite mfg.	Acetone	secondary	2.23e-08	1.13e-09
Ferrite mfg.	Ethylbenzene	secondary	1.94e-08	9.80e-10
Ferrite mfg.	HALON-1301	secondary	1.90e-08	9.62e-10
Steel Prod., cold-rolled, semi-finished	Thallium	secondary	1.78e-08	8.99e-10
Lead	Acetone	secondary	1.72e-08	8.67e-10
Invar	Thorium	secondary	1.65e-08	8.32e-10
Ferrite mfg.	Thorium	secondary	1.61e-08	8.12e-10
Steel Prod., cold-rolled, semi-finished	Zirconium	secondary	1.37e-08	6.94e-10
Lead	Benzo[a]pyrene	secondary	1.23e-08	6.22e-10
Invar	Tin	secondary	9.53e-09	4.82e-10
Invar	Phosphorus pentoxide	secondary	9.39e-09	4.74e-10
Ferrite mfg.	Tin	secondary	9.31e-09	4.70e-10
Ferrite mfg.	Phosphorus pentoxide	secondary	9.16e-09	4.63e-10
Invar	Trichlorofluoromethane	secondary	8.09e-09	4.09e-10
Invar	Hydrogen cyanide	secondary	7.58e-09	3.83e-10
Invar	Hvdrazine	secondary	7.51e-09	3.79e-10
Invar	Scandium	secondary	7.47e-09	3.77e-10
Ferrite mfg.	Hvdrogen cvanide	secondary	7.40e-09	3.74e-10
Ferrite mfg.	Hydrazine	secondary	7.33e-09	3.70e-10
Ferrite mfg.	Scandium	secondary	7.29e-09	3.68e-10
Invar	Perfluoromethane	secondary	5.70e-09	2.88e-10
Steel Prod., cold-rolled, semi-finished	Nitrites	secondary	5.63e-09	2.84e-10
Ferrite mfg.	Perfluoromethane	secondary	5.57e-09	2.81e-10
Steel Prod cold-rolled semi-finished	Chromium (III)	secondary	5 29e-09	2.67e-10
Invar	Thallium	secondary	5 10e-09	2.58e-10
Ferrite mfg	Chromium (III)	secondary	5.01e-09	2.53e-10
Ferrite mfg	Thallium	secondary	4 98e-09	2.55e 10
Invar	Zirconium	secondary	3 94e-09	1.99e-10
Steel Prod cold-rolled semi-finished	Edetic acid (EDTA)	secondary	3.86e-09	1.95e-10
Ferrite mfg	Zirconium	secondary	3.84e-09	1.95 c 10
Invar	Hypochlorous acid	secondary	3.54e-09	1.5 te 10
Steel Prod cold-rolled semi-finished	Hypochlorous acid	secondary	3.52e-09	1.79 c 10
Ferrite mfg	Hypochlorous acid	secondary	3.52e 0) 3.46e-09	1.76e 10
Steel Brod cold rolled semi finished	Cobalt (Co L Co II Co III)	secondary	3.14e.00	1.790-10
Ferrite mfg	Trichlorofluoromethane	secondary	3.01e-09	1.59c-10
ABS Production	HCEC 22	secondary	2.69=.09	1.36e 10
Inver	HEC 125	secondary	2.090-09	1.300-10
Ferrite mfg	HFC-125	secondary	2.506-09	1.100-10
Invor	Chlorine	secondary	2.240-09	1.136-10
Farrita mfa	Chlorine	secondary	2.226-09	1.120-10
Inver	Dichloromethana	secondary	2.176-09	7.610.11
Steel Dred cold rolled court finited	Dichloromethere	secondary	1.516-09	7.010-11
steer Frou., cold-rolled, semi-linished	Diemoromethane	secondary	1.50e-09	7.55e-11
Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
---	-------------------------------	---------------	-------------------------------------	------------
Ferrite mfg.	Dichloromethane	secondary	1.47e-09	7.43e-11
Invar	Dichlorodifluoromethane	secondary	1.44e-09	7.27e-11
Invar	Edetic acid (EDTA)	secondary	1.09e-09	5.52e-11
Ferrite mfg.	Edetic acid (EDTA)	secondary	1.07e-09	5.39e-11
Invar	CFC-13	secondary	1.01e-09	5.09e-11
Steel Prod., cold-rolled, semi-finished	Isopropylpropionate	secondary	4.05e-10	2.04e-11
Ferrite mfg.	Isopropylpropionate	secondary	3.97e-10	2.01e-11
Invar	Halogenated matter (organic)	secondary	2.74e-10	1.39e-11
Ferrite mfg.	Halogenated matter (organic)	secondary	2.68e-10	1.35e-11
Steel Prod., cold-rolled, semi-finished	Lithium salts	secondary	1.98e-10	1.00e-11
Invar	Ethanethiol	secondary	1.74e-10	8.80e-12
Ferrite mfg.	Ethanethiol	secondary	1.70e-10	8.59e-12
Ferrite mfg.	Tin (Sn++, Sn4+)	secondary	1.46e-10	7.36e-12
Steel Prod., cold-rolled, semi-finished	Perfluoromethane	secondary	8.98e-11	4.54e-12
Invar	Lithium salts	secondary	5.61e-11	2.84e-12
Invar	Chloroform	secondary	5.52e-11	2.79e-12
Steel Prod., cold-rolled, semi-finished	Chloroform	secondary	5.48e-11	2.77e-12
Ferrite mfg.	Lithium salts	secondary	5.48e-11	2.77e-12
Ferrite mfg.	Chloroform	secondary	5.39e-11	2.72e-12
Lead	Acrolein	secondary	2.38e-11	1.20e-12
Lead	Benzaldehyde	secondary	1.35e-11	6.81e-13
Invar	Perfluoroethane	secondary	5.35e-12	2.70e-13
Invar	Trichloroethylene	secondary	7.98e-13	4.03e-14
Steel Prod., cold-rolled, semi-finished	Trichloroethylene	secondary	7.91e-13	4.00e-14
Ferrite mfg.	Trichloroethylene	secondary	7.79e-13	3.94e-14
Invar	Propionaldehyde	secondary	2.08e-13	1.05e-14
Steel Prod., cold-rolled, semi-finished	Propionaldehyde	secondary	2.08e-13	1.05e-14
Ferrite mfg.	Propionaldehyde	secondary	2.03e-13	1.03e-14
Steel Prod., cold-rolled, semi-finished	Benzaldehyde	secondary	6.15e-14	3.11e-15
Invar	Benzaldehyde	secondary	6.10e-14	3.08e-15
Ferrite mfg.	Benzaldehyde	secondary	5.95e-14	3.01e-15
Invar	Pentachlorobenzene	secondary	2.77e-14	1.40e-15
Steel Prod., cold-rolled, semi-finished	Tetrachloroethylene	secondary	2.20e-14	1.11e-15
Invar	Tetrachloroethylene	secondary	2.17e-14	1.10e-15
Ferrite mfg.	Tetrachloroethylene	secondary	2.12e-14	1.07e-15
Invar	Acrolein	secondary	9.90e-15	5.00e-16
Ferrite mfg.	1,2-Dichlorotetrafluoroethane	secondary	7.63e-15	3.86e-16
Invar	1,1,1-Trichloroethane	secondary	3.25e-15	1.64e-16
Steel Prod., cold-rolled, semi-finished	1,1,1-Trichloroethane	secondary	3.21e-15	1.62e-16
Ferrite mfg.	1,1,1-Trichloroethane	secondary	3.16e-15	1.60e-16
Invar	Pentachlorophenol	secondary	1.55e-15	7.84e-17
Invar	Hexachloroethane	secondary	1.24e-16	6.27e-18
Steel Prod., cold-rolled, semi-finished	Hexachloroethane	secondary	1.23e-16	6.22e-18
Ferrite mfg.	Hexachloroethane	secondary	1.21e-16	6.12e-18
Ferrite mfg.	Dichlorodifluoromethane	secondary	1.07e-16	5.39e-18
Ferrite mfg.	CFC-13	secondary	7.48e-17	3.78e-18
Invar	HCFC-22	secondary	6.78e-17	3.43e-18
Ferrite mfg.	HCFC-22	secondary	6.62e-17	3.35e-18
Total Materials Processing			2.28e+02	1.15e+01
Manufacturing Life-cycle Stage			1.200 . 02	

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Japanese Electric Grid	Sulfur dioxide	model/secondary	5.69e+01	2.88e+00
US electric grid	Sulfur dioxide	model/secondary	2.63e+01	1.33e+00
LPG Production	Carbon monoxide	secondary	5.77e+00	2.92e-01
LPG Production	Methane	secondary	1.68e+00	8.48e-02
LPG Production	Sulfur oxides	secondary	1.60e+00	8.10e-02
LPG Production	Nitrogen oxides	secondary	1.15e+00	5.79e-02
LPG Production	Vanadium	secondary	1.05e+00	5.29e-02
LPG Production	Benzene	secondary	8.58e-01	4.34e-02
Glass/frit	Fluorides (F-)	primary	5.85e-01	2.96e-02
LPG Production	PM	secondary	2.58e-01	1.30e-02
CRT tube mfg.	Carbon monoxide	primary	2.29e-01	1.16e-02
LPG Production	Arsenic	secondary	2.06e-01	1.04e-02
LPG Production	Formaldehyde	secondary	1.34e-01	6.77e-03
Natural Gas Prod.	Methane	secondary	1.00e-01	5.07e-03
Glass/frit	Nitrogen oxides	primary	8.83e-02	4.46e-03
Japanese Electric Grid	Nitrogen oxides	model/secondary	8.13e-02	4.11e-03
Natural Gas Prod.	Benzene	secondary	7.57e-02	3.83e-03
Natural Gas Prod.	Carbon monoxide	secondary	6.87e-02	3.47e-03
CRT monitor assembly	Tricresyl phosphate	primary	4.59e-02	2.32e-03
CRT tube mfg.	Phosphorus (yellow or white)	primary	4.00e-02	2.02e-03
Japanese Electric Grid	Carbon monoxide	model/secondary	3.78e-02	1.91e-03
US electric grid	Nitrogen oxides	model/secondary	3.76e-02	1.90e-03
LPG Production	Hydrochloric acid	secondary	3.68e-02	1.86e-03
LPG Production	Nitrous oxide	secondary	3.24e-02	1.64e-03
Fuel Oil #6 Prod.	Carbon monoxide	secondary	2.83e-02	1.43e-03
Japanese Electric Grid	Vanadium	model/secondary	2.27e-02	1.15e-03
US electric grid	Methane	model/secondary	2.06e-02	1.04e-03
CRT tube mfg.	Sulfur oxides	primary	1.99e-02	1.01e-03
Natural Gas Prod.	Nitrogen oxides	secondary	1.86e-02	9.42e-04
US electric grid	Carbon monoxide	model/secondary	1.74e-02	8.80e-04
Fuel Oil #2 Prod.	Carbon monoxide	secondary	1.68e-02	8.50e-04
Fuel Oil #6 Prod.	Methane	secondary	1.18e-02	5.95e-04
Fuel Oil #6 Prod.	Sulfur oxides	secondary	1.09e-02	5.53e-04
Fuel Oil #6 Prod.	Vanadium	secondary	1.08e-02	5.44e-04
Japanese Electric Grid	Arsenic	model/secondary	1.04e-02	5.27e-04
LPG Production	Phosphorus (yellow or white)	secondary	9.81e-03	4.96e-04
US electric grid	Arsenic	model/secondary	8.91e-03	4.50e-04
Fuel Oil #6 Prod.	Nitrogen oxides	secondary	8.19e-03	4.14e-04
US electric grid	Hydrochloric acid	model/secondary	7.87e-03	3.98e-04
LPG Production	Fluorides (F-)	secondary	7.54e-03	3.81e-04
LPG Production	Selenium	secondary	7.41e-03	3.75e-04
CRT tube mfg.	Fluoride	primary	6.91e-03	3.49e-04
Japanese Electric Grid	Hydrochloric acid	model/secondary	6.28e-03	3.17e-04
LPG Production	Hydrogen sulfide	secondary	6.13e-03	3.10e-04
LPG Production	Ammonia	secondary	5.91e-03	2.99e-04
Glass/frit	Fluorides (F-)	primary	5.85e-03	2.95e-04
Fuel Oil #2 Prod.	Methane	secondary	5.14e-03	2.60e-04
Fuel Oil #6 Prod.	Benzene	secondary	4.90e-03	2.48e-04
Fuel Oil #2 Prod.	Sulfur oxides	secondary	4.87e-03	2.46e-04
PWB Mfg.	Formaldehvde	model/secondary	4.45e-03	2.25e-04
U		· · · · · · · · · · · · · · · · · · ·		

Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
CRT tube mfg.	Nitrogen oxides	primary	4.34e-03	2.19e-04
Japanese Electric Grid	PM-10	model/secondary	4.00e-03	2.02e-04
Fuel Oil #2 Prod.	Nitrogen oxides	secondary	3.51e-03	1.78e-04
Fuel Oil #2 Prod.	Vanadium	secondary	3.45e-03	1.74e-04
CRT tube mfg.	Dimethyl Formamide	primary	3.07e-03	1.55e-04
Fuel Oil #2 Prod.	Benzene	secondary	2.55e-03	1.29e-04
Glass/frit	Carbon monoxide	primary	2.16e-03	1.09e-04
LPG Production	Hydrofluoric acid	secondary	2.01e-03	1.01e-04
Fuel Oil #6 Prod.	PM	secondary	1.85e-03	9.34e-05
US electric grid	PM-10	model/secondary	1.84e-03	9.32e-05
Fuel Oil #6 Prod.	Arsenic	secondary	1.72e-03	8.72e-05
CRT tube mfg.	Zinc (elemental)	primary	1.65e-03	8.33e-05
LPG Production	Molybdenum	secondary	1.65e-03	8.33e-05
LPG Production	Chromium (VI)	secondary	1.58e-03	7.97e-05
LPG Production	Ethane	secondary	1.55e-03	7.84e-05
Fuel Oil #4 Prod.	Carbon monoxide	secondary	1.52e-03	7.66e-05
US electric grid	Selenium	model/secondary	1.48e-03	7.48e-05
Fuel Oil #6 Prod.	Formaldehyde	secondary	1.37e-03	6.92e-05
Japanese Electric Grid	Fluorides (F-)	model/secondary	1.32e-03	6.65e-05
Japanese Electric Grid	Selenium	model/secondary	1.27e-03	6.44e-05
LPG Production	Zinc (elemental)	secondary	1.20e-03	6.06e-05
Japanese Electric Grid	Formaldehyde	model/secondary	1.17e-03	5.91e-05
US electric grid	Vanadium	model/secondary	1.11e-03	5.63e-05
Glass/frit	Nitrogen oxides	primary	1.07e-03	5.39e-05
LPG Production	Hexane	secondary	9.02e-04	4.56e-05
Natural Gas Prod.	PM	secondary	8.28e-04	4.18e-05
Fuel Oil #2 Prod.	PM	secondary	7.92e-04	4.00e-05
LPG Production	Phenol	secondary	7.15e-04	3.61e-05
Fuel Oil #2 Prod.	Arsenic	secondary	6.52e-04	3.29e-05
LPG Production	Pentane	secondary	6.51e-04	3.29e-05
CRT tube mfg.	Toluene	primary	6.41e-04	3.24e-05
Japanese Electric Grid	Zinc (elemental)	model/secondary	6.13e-04	3.10e-05
Natural Gas Prod.	Sulfur oxides	secondary	5.51e-04	2.79e-05
Fuel Oil #4 Prod.	Methane	secondary	5.21e-04	2.63e-05
Fuel Oil #4 Prod.	Sulfur oxides	secondary	4.90e-04	2.48e-05
LPG Production	PM-10	secondary	4.50e-04	2.27e-05
Fuel Oil #2 Prod.	Formaldehyde	secondary	4.41e-04	2.23e-05
US electric grid	Hydrofluoric acid	model/secondary	4.29e-04	2.17e-05
Natural Gas Prod.	Ammonia	secondary	4.26e-04	2.15e-05
LPG Production	Nickel	secondary	4.12e-04	2.08e-05
Fuel Oil #4 Prod.	Vanadium	secondary	4.03e-04	2.04e-05
Fuel Oil #4 Prod.	Nitrogen oxides	secondary	3.59e-04	1.81e-05
Japanese Electric Grid	Hydrofluoric acid	model/secondary	3.43e-04	1.73e-05
Japanese Electric Grid	Antimony	model/secondary	3.24e-04	1.64e-05
Natural Gas Prod.	Arsenic	secondary	3.15e-04	1.59e-05
Fuel Oil #6 Prod.	Hydrochloric acid	secondary	2.97e-04	1.50e-05
Fuel Oil #6 Prod.	Nitrous oxide	secondary	2.90e-04	1.47e-05
LPG Production	Aluminum (+3)	secondary	2.83e-04	1.43e-05
CRT tube mfg.	Nickel	primary	2.68e-04	1.35e-05
Fuel Oil #4 Prod.	Benzene	secondary	2.41e-04	1.22e-05

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Japanese Electric Grid	Nitrous oxide	model/secondary	2.24e-04	1.13e-05
Glass/frit	PM	primary	2.18e-04	1.10e-05
LPG Production	Antimony	secondary	2.14e-04	1.08e-05
Japanese Electric Grid	Methane	model/secondary	1.63e-04	8.22e-06
US electric grid	Formaldehyde	model/secondary	1.55e-04	7.82e-06
Japanese Electric Grid	Molybdenum	model/secondary	1.32e-04	6.65e-06
LPG Production	Nitrate	secondary	1.21e-04	6.12e-06
Fuel Oil #2 Prod.	Hydrochloric acid	secondary	1.15e-04	5.83e-06
US electric grid	Benzene	model/secondary	1.11e-04	5.63e-06
US electric grid	Nitrous oxide	model/secondary	1.08e-04	5.48e-06
Fuel Oil #2 Prod.	Nitrous oxide	secondary	1.04e-04	5.24e-06
CRT tube mfg.	Molybdenum	primary	1.02e-04	5.18e-06
Glass/frit	Sulfur oxides	primary	1.02e-04	5.14e-06
Fuel Oil #6 Prod.	Phosphorus (yellow or white)	secondary	9.77e-05	4.94e-06
Japanese Electric Grid	Benzene	model/secondary	9.11e-05	4.60e-06
Glass/frit	Lead	primary	8.74e-05	4.42e-06
Fuel Oil #4 Prod.	PM	secondary	8.09e-05	4.09e-06
Fuel Oil #4 Prod.	Arsenic	secondary	7.04e-05	3.56e-06
Natural Gas Prod.	Vanadium	secondary	6.96e-05	3.52e-06
Natural Gas Prod.	Hydrochloric acid	secondary	6.39e-05	3.23e-06
Fuel Oil #6 Prod.	Hydrogen sulfide	secondary	6.35e-05	3.21e-06
US electric grid	Phosphorus (yellow or white)	model/secondary	6.27e-05	3.17e-06
Fuel Oil #6 Prod.	Selenium	secondary	6.07e-05	3.07e-06
Glass/frit	Carbon monoxide	primary	6.03e-05	3.05e-06
Fuel Oil #6 Prod.	Fluorides (F-)	secondary	5.86e-05	2.96e-06
Fuel Oil #4 Prod.	Formaldehyde	secondary	5.14e-05	2.60e-06
Japanese Electric Grid	Nickel	model/secondary	4.98e-05	2.52e-06
CRT tube mfg.	Copper	primary	4.05e-05	2.04e-06
LPG Production	Copper	secondary	3.49e-05	1.76e-06
LPG Production	Methyl hydrazine	secondary	3.44e-05	1.74e-06
LPG Production	Silicon	secondary	3.38e-05	1.71e-06
Glass/frit	Nitrates/nitrites	primary	3.34e-05	1.69e-06
LPG Production	2-Chloroacetophenone	secondary	3.22e-05	1.63e-06
Fuel Oil #2 Prod.	Phosphorus (yellow or white)	secondary	3.21e-05	1.62e-06
LPG Production	Dimethylbenzanthracene	secondary	3.19e-05	1.61e-06
LPG Production	Bromomethane	secondary	3.18e-05	1.61e-06
Fuel Oil #6 Prod.	Ammonia	secondary	3.18e-05	1.61e-06
Japanese Electric Grid	Barium	model/secondary	2.94e-05	1.48e-06
US electric grid	Zinc (elemental)	model/secondary	2.89e-05	1.46e-06
Natural Gas Prod.	Formaldehyde	secondary	2.77e-05	1.40e-06
LPG Production	Naphthalene	secondary	2.61e-05	1.32e-06
CRT tube mfg.	Xylene (mixed isomers)	primary	2.55e-05	1.29e-06
LPG Production	Lead	secondary	2.47e-05	1.25e-06
US electric grid	Antimony	model/secondary	2.37e-05	1.20e-06
Fuel Oil #2 Prod.	Fluorides (F-)	secondary	2.35e-05	1.19e-06
Fuel Oil #2 Prod.	Selenium	secondary	2.33e-05	1.18e-06
Natural Gas Prod.	Nitrous oxide	secondary	2.31e-05	1,17e-06
LPG Production	Manganese	secondary	2.23e-05	1.13e-06
LPG Production	Bervllium	secondary	2.15e-05	1.08e-06
Fuel Oil #2 Prod.	Hydrogen sulfide	secondary	2.02e-05	1.02e-06
			, 00	

 Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
LPG Production	Barium	secondary	1.86e-05	9.42e-07
Fuel Oil #2 Prod.	Ammonia	secondary	1.74e-05	8.81e-07
LPG Production	Cyanide (-1)	secondary	1.67e-05	8.45e-07
Japanese Electric Grid	Naphthalene	model/secondary	1.66e-05	8.37e-07
Fuel Oil #6 Prod.	Hydrofluoric acid	secondary	1.62e-05	8.19e-07
LPG Production	Barium cmpds	secondary	1.59e-05	8.02e-07
Fuel Oil #6 Prod.	Molybdenum	secondary	1.56e-05	7.88e-07
Natural Gas Prod.	Ethane	secondary	1.40e-05	7.06e-07
Natural Gas Prod.	Fluorides (F-)	secondary	1.36e-05	6.87e-07
Fuel Oil #4 Prod.	Hydrochloric acid	secondary	1.23e-05	6.22e-07
Natural Gas Prod.	Selenium	secondary	1.21e-05	6.13e-07
Fuel Oil #4 Prod.	Nitrous oxide	secondary	1.15e-05	5.81e-07
Japanese Electric Grid	Chromium (VI)	model/secondary	9.99e-06	5.05e-07
LPG Production	Cadmium	secondary	9.21e-06	4.65e-07
Fuel Oil #6 Prod.	Ethane	secondary	8.87e-06	4.48e-07
US electric grid	Chromium (VI)	model/secondary	8.58e-06	4.33e-07
Natural Gas Prod.	Zinc (elemental)	secondary	8.50e-06	4.30e-07
Natural Gas Prod.	Hexane	secondary	8.11e-06	4.10e-07
US electric grid	Molybdenum	model/secondary	7.93e-06	4.01e-07
US electric grid	Methyl hydrazine	model/secondary	7.36e-06	3.72e-07
Japanese Electric Grid	Copper	model/secondary	7.14e-06	3.61e-07
US electric grid	2-Chloroacetophenone	model/secondary	6.89e-06	3.48e-07
LPG Production	Carbon disulfide	secondary	6.85e-06	3.46e-07
US electric grid	Bromomethane	model/secondary	6.81e-06	3.44e-07
Fuel Oil #6 Prod.	Zinc (elemental)	secondary	6.59e-06	3.33e-07
Fuel Oil #6 Prod.	Chromium (VI)	secondary	6.39e-06	3.23e-07
Fuel Oil #2 Prod.	Hydrofluoric acid	secondary	6.30e-06	3.18e-07
CRT tube mfg.	Lead	primary	6.03e-06	3.05e-07
Japanese Electric Grid	Methyl hydrazine	model/secondary	5.88e-06	2.97e-07
Natural Gas Prod.	Pentane	secondary	5.86e-06	2.96e-07
Japanese Electric Grid	2-Chloroacetophenone	model/secondary	5.51e-06	2.78e-07
Japanese Electric Grid	Bromomethane	model/secondary	5.44e-06	2.75e-07
Fuel Oil #2 Prod.	Molybdenum	secondary	5.33e-06	2.69e-07
LPG Production	Benzyl chloride	secondary	5.24e-06	2.65e-07
Fuel Oil #6 Prod.	Hexane	secondary	5.16e-06	2.61e-07
LPG Production	Cobalt	secondary	5.02e-06	2.54e-07
Fuel Oil #6 Prod.	PM-10	secondary	4.66e-06	2.36e-07
Fuel Oil #2 Prod.	Ethane	secondary	4.61e-06	2.33e-07
Fuel Oil #6 Prod.	Nickel	secondary	3.99e-06	2.02e-07
LPG Production	Aluminum (elemental)	secondary	3.87e-06	1.96e-07
Fuel Oil #6 Prod.	Pentane	secondary	3.72e-06	1.88e-07
Fuel Oil #4 Prod.	Phosphorus (yellow or white)	secondary	3.70e-06	1.87e-07
US electric grid	Nickel	model/secondary	3.70e-06	1.87e-07
LPG Production	Chloroform	secondary	3.70e-06	1.87e-07
CRT tube mfg.	Manganese	primary	3.60e-06	1.82e-07
US electric grid	Cyanide (-1)	model/secondary	3.58e-06	1.81e-07
Fuel Oil #2 Prod.	Zinc (elemental)	secondary	3.54e-06	1.79e-07
Natural Gas Prod.	Hydrofluoric acid	secondary	3.49e-06	1.76e-07
LPG Production	Propionaldehyde	secondary	3.42e-06	1.73e-07
Japanese Electric Grid	Cyanide (-1)	model/secondary	2.86e-06	1.44e-07

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
US electric grid	2,3,7,8-TCDD	model/secondary	2.71e-06	1.37e-07
Fuel Oil #2 Prod.	Hexane	secondary	2.68e-06	1.35e-07
Fuel Oil #4 Prod.	Selenium	secondary	2.50e-06	1.26e-07
Fuel Oil #2 Prod.	Chromium (VI)	secondary	2.48e-06	1.25e-07
Fuel Oil #4 Prod.	Fluorides (F-)	secondary	2.47e-06	1.25e-07
Fuel Oil #6 Prod.	Phenol	secondary	2.42e-06	1.22e-07
Fuel Oil #4 Prod.	Hydrogen sulfide	secondary	2.37e-06	1.20e-07
Japanese Electric Grid	Cobalt	model/secondary	2.37e-06	1.20e-07
Natural Gas Prod.	Molybdenum	secondary	2.31e-06	1.17e-07
Japanese Electric Grid	2,3,7,8-TCDD	model/secondary	2.23e-06	1.12e-07
US electric grid	Naphthalene	model/secondary	2.19e-06	1.10e-07
Fuel Oil #2 Prod.	Phenol	secondary	2.01e-06	1.01e-07
LPG Production	Acrolein	secondary	1.94e-06	9.80e-08
Fuel Oil #2 Prod.	Pentane	secondary	1.93e-06	9.77e-08
US electric grid	Barium	model/secondary	1.86e-06	9.39e-08
Natural Gas Prod.	Phosphorus (yellow or white)	secondary	1.77e-06	8.94e-08
Fuel Oil #4 Prod.	Ammonia	secondary	1.62e-06	8.17e-08
LPG Production	Methyl chloride	secondary	1.61e-06	8.13e-08
Japanese Electric Grid	Cadmium	model/secondary	1.60e-06	8.09e-08
Fuel Oil #2 Prod.	PM-10	secondary	1.48e-06	7.50e-08
Natural Gas Prod.	Chromium (VI)	secondary	1.48e-06	7.47e-08
US electric grid	Carbon disulfide	model/secondary	1.46e-06	7.40e-08
Fuel Oil #2 Prod.	Nickel	secondary	1.34e-06	6.77e-08
Glass/frit	РМ	primary	1.33e-06	6.74e-08
Fuel Oil #6 Prod.	Antimony	secondary	1.19e-06	6.00e-08
Japanese Electric Grid	Carbon disulfide	model/secondary	1.17e-06	5.91e-08
LPG Production	Di(2-ethylhexyl)phthalate	secondary	1 16e-06	5.86e-08
US electric grid	Benzyl chloride	model/secondary	1 12e-06	5 67e-08
Fuel Oil #6 Prod	Aluminum (+3)	secondary	9 58e-07	4 84e-08
LPG Production	Acetaldehyde	secondary	9 23e-07	4 67e-08
Japanese Electric Grid	Benzyl chloride	model/secondary	8 95e-07	4 52e-08
US electric grid	Cadmium	model/secondary	8.74e-07	4 42e-08
LPG Production	Mercury	secondary	8.5 fe-07	4 39e-08
Fuel Oil #2 Prod	Aluminum (+3)	secondary	7 95e-07	4 02e-08
US electric grid	Chloroform	model/secondary	7.93e-07	4 00e-08
US electric grid	Propionaldehyde	model/secondary	7.91e-07	3.69e-08
US electric grid	Manganese	model/secondary	7.51e 07	3.68e-08
I PG Production	Cadmium cmpds	secondary	6.95e-07	3.51e-08
Fuel Oil #4 Prod	Hydrofluoric acid	secondary	6.72e-07	3.39e-08
I PG Production	Dimethyl sulfate	secondary	6.42e-07	3.25e-08
End Oil #2 Prod	Antimony	secondary	6 3/10 07	3 20e 08
Japanese Electric Grid	Chloroform	model/secondem/	6.310.07	3.200-08
US electric grid	Fluoride	model/secondary	6 23e 07	3.15e.08
L PG Production	Teluene	socondary	6 230 07	3.150.08
CPT tube mfg	Cyanida (1)	nrimory	6.062.07	3.156-08
Evel Oil #4 Prod	Cyaillut (-1) Molyhdenum	primary	6.02c.07	3.050.09
Japapasa Elastria Grid	Dronionaldahyda	model/secondem-	5.94-07	2.05-08
Gloss/frit	Zing (alemental)	niouel/secondary	5.846-07	2.950-08
Japapasa Flastria Grid	Zine (cieniental)	primary model/secondem:	5.55e-07	2.810-08
		model/secondary	4.94e-07	2.500-08
ruei Oli #0 Prod.	Initrate	secondary	4.85e-07	2.45e-08

Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
US electric grid	Beryllium	model/secondary	4.65e-07	2.35e-08
Natural Gas Prod.	Antimony	secondary	4.46e-07	2.25e-08
Glass/frit	Chromium	primary	4.41e-07	2.23e-08
Fuel Oil #4 Prod.	Ethane	secondary	4.36e-07	2.20e-08
US electric grid	Acrolein	model/secondary	4.15e-07	2.10e-08
US electric grid	Lead	model/secondary	4.07e-07	2.06e-08
US electric grid	Cobalt	model/secondary	3.88e-07	1.96e-08
LPG Production	1,4-Dichlorobenzene	secondary	3.62e-07	1.83e-08
US electric grid	Copper	model/secondary	3.57e-07	1.80e-08
Fuel Oil #6 Prod.	Silicon	secondary	3.50e-07	1.77e-08
Fuel Oil #2 Prod.	Nitrate	secondary	3.45e-07	1.74e-08
US electric grid	Methyl chloride	model/secondary	3.44e-07	1.74e-08
Japanese Electric Grid	Acrolein	model/secondary	3.31e-07	1.67e-08
Fuel Oil #4 Prod.	Zinc (elemental)	secondary	3.31e-07	1.67e-08
Fuel Oil #6 Prod.	Copper	secondary	3.17e-07	1.60e-08
LPG Production	Isophorone	secondary	3.10e-07	1.57e-08
Natural Gas Prod.	Dimethylbenzanthracene	secondary	2.87e-07	1.45e-08
LPG Production	Aromatic hydrocarbons	secondary	2.80e-07	1.42e-08
Fuel Oil #6 Prod.	Methyl hydrazine	secondary	2.78e-07	1.40e-08
Glass/frit	Nickel	primary	2.77e-07	1.40e-08
Japanese Electric Grid	Methyl chloride	model/secondary	2.75e-07	1.39e-08
Fuel Oil #4 Prod.	Chromium (VI)	secondary	2.64e-07	1.34e-08
Fuel Oil #6 Prod.	2-Chloroacetophenone	secondary	2.60e-07	1.31e-08
Fuel Oil #6 Prod.	Bromomethane	secondary	2.57e-07	1.30e-08
Fuel Oil #4 Prod.	Hexane	secondary	2.53e-07	1.28e-08
LPG Production	Methyl ethyl ketone	secondary	2.48e-07	1.26e-08
US electric grid	Di(2-ethylhexyl)phthalate	model/secondary	2.48e-07	1.25e-08
LPG Production	Tetrachloroethylene	secondary	2.46e-07	1.24e-08
Japanese Electric Grid	Toluene	model/secondary	2.39e-07	1.21e-08
LPG Production	1,2-Dichloroethane	secondary	2.18e-07	1.10e-08
LPG Production	Methyl methacrylate	secondary	2.12e-07	1.07e-08
Fuel Oil #6 Prod.	Lead	secondary	2.00e-07	1.01e-08
Japanese Electric Grid	Di(2-ethylhexyl)phthalate	model/secondary	1.98e-07	1.00e-08
US electric grid	Acetaldehyde	model/secondary	1.98e-07	9.98e-09
US electric grid	Hexane	model/secondary	1.92e-07	9.69e-09
LPG Production	Styrene	secondary	1.88e-07	9.48e-09
Fuel Oil #4 Prod.	Pentane	secondary	1.83e-07	9.24e-09
Fuel Oil #6 Prod.	Dimethylbenzanthracene	secondary	1.82e-07	9.22e-09
Fuel Oil #6 Prod.	Manganese	secondary	1.79e-07	9.04e-09
LPG Production	Bromoform	secondary	1 75e-07	8 85e-09
Natural Gas Prod	Nickel	secondary	1.74e-07	8 81e-09
Fuel Oil #4 Prod	PM-10	secondary	1.74e-07	8 79e-09
LPG Production	Dichloromethane	secondary	1.73e-07	8.74e-09
LPG Production	Chromium (III)	secondary	1.75e-07	8.67e-09
Fuel Oil #6 Prod	Beryllium	secondary	1.72e-07	8.64e-09
Fuel Oil #4 Prod	Phenol	secondary	1.63e-07	8 24e-09
Fuel Oil #6 Prod	Nanhthalene	secondary	1.55c-07	8 00e-09
Japanese Electric Grid	Acetaldehyde	model/secondary	1.58e-07	7.97e-00
Jananese Electric Grid	Hexane	model/secondary	1.53e-07	7 74e-09
Fuel Oil #4 Prod	Nickel	secondary	1.536-07	7.730-00
I PG Production	Chlorobenzene	secondary	1.550-07	7.08e-09
		secondary	1.400-07	1.000-09

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
US electric grid	Dimethyl sulfate	model/secondary	1.37e-07	6.94e-09
Fuel Oil #6 Prod.	Cyanide (-1)	secondary	1.35e-07	6.82e-09
Natural Gas Prod.	Hydrogen sulfide	secondary	1.35e-07	6.80e-09
Natural Gas Prod.	Naphthalene	secondary	1.32e-07	6.66e-09
LPG Production	Copper (+1 & +2)	secondary	1.31e-07	6.62e-09
US electric grid	Mercury	model/secondary	1.20e-07	6.06e-09
Japanese Electric Grid	Mercury	model/secondary	1.18e-07	5.97e-09
LPG Production	2,4-Dinitrotoluene	secondary	1.12e-07	5.68e-09
Fuel Oil #2 Prod.	Copper	secondary	1.12e-07	5.65e-09
Fuel Oil #2 Prod.	Silicon	secondary	1.12e-07	5.64e-09
Japanese Electric Grid	Dimethyl sulfate	model/secondary	1.10e-07	5.54e-09
Fuel Oil #2 Prod.	Methyl hydrazine	secondary	1.08e-07	5.45e-09
LPG Production	Acetophenone	secondary	1.03e-07	5.21e-09
Fuel Oil #2 Prod.	2-Chloroacetophenone	secondary	1.01e-07	5.11e-09
Fuel Oil #2 Prod.	Bromomethane	secondary	9.99e-08	5.05e-09
Fuel Oil #6 Prod.	Barium	secondary	9.97e-08	5.04e-09
Fuel Oil #2 Prod.	Dimethylbenzanthracene	secondary	9.48e-08	4.79e-09
Fuel Oil #2 Prod.	Naphthalene	secondary	7.83e-08	3.95e-09
Fuel Oil #2 Prod.	Lead	secondary	7.76e-08	3.92e-09
Fuel Oil #6 Prod.	Cadmium	secondary	7.24e-08	3.66e-09
Fuel Oil #2 Prod.	Manganese	secondary	7.00e-08	3.53e-09
LPG Production	o-xylene	secondary	6.86e-08	3.47e-09
US electric grid	Toluene	model/secondary	6.78e-08	3.43e-09
Fuel Oil #2 Prod.	Beryllium	secondary	6.72e-08	3.39e-09
US electric grid	Isophorone	model/secondary	6.64e-08	3.36e-09
Fuel Oil #4 Prod.	Aluminum (+3)	secondary	6.46e-08	3.26e-09
Natural Gas Prod.	Methyl hydrazine	secondary	5.97e-08	3.01e-09
Fuel Oil #4 Prod.	Antimony	secondary	5.94e-08	3.00e-09
Natural Gas Prod.	Barium	secondary	5.90e-08	2.98e-09
Natural Gas Prod.	2-Chloroacetophenone	secondary	5.59e-08	2.82e-09
LPG Production	Ethylbenzene	secondary	5.56e-08	2.81e-09
Natural Gas Prod.	Bromomethane	secondary	5.52e-08	2.79e-09
Fuel Oil #6 Prod.	Carbon disulfide	secondary	5.52e-08	2.79e-09
Fuel Oil #2 Prod.	Barium	secondary	5.49e-08	2.77e-09
Fuel Oil #6 Prod.	Barium cmpds	secondary	5.37e-08	2.71e-09
US electric grid	Methyl ethyl ketone	model/secondary	5.32e-08	2.69e-09
Japanese Electric Grid	Isophorone	model/secondary	5.30e-08	2.68e-09
Glass/frit	Barium	primary	5.29e-08	2.67e-09
Natural Gas Prod.	Copper	secondary	5.28e-08	2.67e-09
US electric grid	Tetrachloroethylene	model/secondary	5.25e-08	2.65e-09
Fuel Oil #2 Prod.	Cyanide (-1)	secondary	5.25e-08	2.65e-09
US electric grid	1,2-Dichloroethane	model/secondary	4.67e-08	2.36e-09
US electric grid	Methyl methacrylate	model/secondary	4.54e-08	2.30e-09
Fuel Oil #2 Prod.	Barium cmpds	secondary	4.45e-08	2.25e-09
Natural Gas Prod.	Lead	secondary	4.33e-08	2.19e-09
Natural Gas Prod.	Beryllium	secondary	4.27e-08	2.16e-09
Fuel Oil #6 Prod.	Cobalt	secondary	4.25e-08	2.14e-09
Japanese Electric Grid	Methyl ethyl ketone	model/secondary	4.24e-08	2.14e-09
Fuel Oil #6 Prod.	Benzyl chloride	secondary	4.23e-08	2.14e-09
Japanese Electric Grid	Tetrachloroethylene	model/secondary	4.19e-08	2.12e-09

Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
US electric grid	Styrene	model/secondary	4.01e-08	2.03e-09
Fuel Oil #6 Prod.	Aluminum (elemental)	secondary	4.01e-08	2.03e-09
Natural Gas Prod.	Manganese	secondary	3.95e-08	2.00e-09
US electric grid	Bromoform	model/secondary	3.75e-08	1.89e-09
Japanese Electric Grid	1,2-Dichloroethane	model/secondary	3.73e-08	1.88e-09
US electric grid	Dichloromethane	model/secondary	3.70e-08	1.87e-09
Japanese Electric Grid	Methyl methacrylate	model/secondary	3.63e-08	1.83e-09
Japanese Electric Grid	Styrene	model/secondary	3.20e-08	1.62e-09
Natural Gas Prod.	Nitrate	secondary	3.01e-08	1.52e-09
US electric grid	Chlorobenzene	model/secondary	3.00e-08	1.51e-09
Japanese Electric Grid	Bromoform	model/secondary	2.99e-08	1.51e-09
Fuel Oil #6 Prod.	Chloroform	secondary	2.98e-08	1.51e-09
Japanese Electric Grid	Dichloromethane	model/secondary	2.95e-08	1.49e-09
Fuel Oil #4 Prod.	Nitrate	secondary	2.93e-08	1.48e-09
Natural Gas Prod.	Cyanide (-1)	secondary	2.90e-08	1.47e-09
Fuel Oil #2 Prod.	Cadmium	secondary	2.88e-08	1.45e-09
LPG Production	Methyl tert-butyl ether	secondary	2.79e-08	1.41e-09
Fuel Oil #6 Prod.	Propionaldehyde	secondary	2.76e-08	1.39e-09
US electric grid	2,4-Dinitrotoluene	model/secondary	2.40e-08	1.21e-09
Japanese Electric Grid	Chlorobenzene	model/secondary	2.39e-08	1.21e-09
LPG Production	Zinc (+2)	secondary	2.35e-08	1.19e-09
LPG Production	Phenanthrene	secondary	2.27e-08	1.15e-09
US electric grid	Acetophenone	model/secondary	2.21e-08	1.12e-09
Fuel Oil #2 Prod.	Carbon disulfide	secondary	2.15e-08	1.09e-09
LPG Production	Vinvl acetate	secondary	1.98e-08	1.00e-09
Japanese Electric Grid	Cumene hydroperoxide	model/secondary	1.95e-08	9.84e-10
LPG Production	3-Methylcholanthrene	secondary	1.92e-08	9.70e-10
Japanese Electric Grid	2.4-Dinitrotoluene	model/secondary	1.91e-08	9.67e-10
LPG Production	Ethylene dibromide	secondary	1.84e-08	9.30e-10
Japanese Electric Grid	Acetophenone	model/secondary	1.76e-08	8.91e-10
Fuel Oil #2 Prod.	Benzyl chloride	secondary	1.64e-08	8.31e-10
Natural Gas Prod.	Cadmium	secondary	1.60e-08	8.06e-10
Fuel Oil #2 Prod.	Cobalt	secondary	1.59e-08	8.03e-10
Fuel Oil #6 Prod.	Acrolein	secondary	1.57e-08	7.91e-10
Natural Gas Prod	Phenol	secondary	1 46e-08	7 37e-10
Glass/frit	Copper	primary	1.42e-08	7.18e-10
Fuel Oil #4 Prod.	Silicon	secondary	1.31e-08	6.61e-10
Fuel Oil #6 Prod	Methyl chloride	secondary	1 30e-08	6 56e-10
Fuel Oil #2 Prod.	Aluminum (elemental)	secondary	1.28e-08	6.45e-10
Fuel Oil #4 Prod	Copper	secondary	1.25e-08	6 30e-10
LPG Production	2-Methylnaphthalene	secondary	1.20e-08	6.07e-10
Natural Gas Prod	Carbon disulfide	secondary	1 19e-08	6.00e-10
US electric grid	Ethylbenzene	model/secondary	1 18e-08	5.97e-10
Fuel Oil #2 Prod	Chloroform	secondary	1 16e-08	5.86e-10
Fuel Oil #4 Prod	Methyl hydrazine	secondary	1 15e-08	5.81e-10
Fuel Oil #4 Prod.	2-Chloroacetophenone	secondary	1.08e-08	5.45e-10
Fuel Oil #2 Prod	Propionaldehyde	secondary	1.000-08 1.07e-08	5.41e-10
Fuel Oil #4 Prod	Bromomethane	secondary	1.07e-08	5 38e-10
LPG Production	Chlorine	secondary	1.07e-08	5 38e-10
Jananese Electric Grid	Ethylbenzene	model/secondary	1.040-08	5.25e-10
Natural Gas Prod	PM-10	secondary	0 880 00	4 996-10
1 maratal 005 1 100.	1 1/1 10	secondary	2.000-09	

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Fuel Oil #6 Prod.	Di(2-ethylhexyl)phthalate	secondary	9.36e-09	4.73e-10
Natural Gas Prod.	Benzyl chloride	secondary	9.09e-09	4.59e-10
Fuel Oil #4 Prod.	Dimethylbenzanthracene	secondary	8.97e-09	4.53e-10
Natural Gas Prod.	Toluene	secondary	8.63e-09	4.36e-10
Fuel Oil #4 Prod.	Lead	secondary	8.28e-09	4.18e-10
LPG Production	1,1,1-Trichloroethane	secondary	7.57e-09	3.82e-10
Fuel Oil #4 Prod.	Naphthalene	secondary	7.55e-09	3.81e-10
Fuel Oil #6 Prod.	Acetaldehyde	secondary	7.45e-09	3.76e-10
Fuel Oil #4 Prod.	Manganese	secondary	7.44e-09	3.76e-10
Fuel Oil #4 Prod.	Beryllium	secondary	7.13e-09	3.60e-10
LPG Production	Acenaphthene	secondary	7.03e-09	3.55e-10
Fuel Oil #6 Prod.	Mercury	secondary	6.94e-09	3.51e-10
Natural Gas Prod.	Chloroform	secondary	6.41e-09	3.24e-10
Fuel Oil #2 Prod.	Acrolein	secondary	6.09e-09	3.08e-10
US electric grid	Methyl tert-butyl ether	model/secondary	5.96e-09	3.01e-10
Natural Gas Prod.	Propionaldehyde	secondary	5.93e-09	2.99e-10
Japanese Electric Grid	Acenaphthene	model/secondary	5.79e-09	2.92e-10
Natural Gas Prod.	Aluminum (+3)	secondary	5.72e-09	2.89e-10
Fuel Oil #4 Prod.	Cyanide (-1)	secondary	5.60e-09	2.83e-10
LPG Production	Ethyl Chloride	secondary	5.36e-09	2.71e-10
Natural Gas Prod.	Cobalt	secondary	5.28e-09	2.67e-10
Japanese Electric Grid	Phenanthrene	model/secondary	5.21e-09	2.63e-10
Fuel Oil #6 Prod.	Dimethyl sulfate	secondary	5.18e-09	2.62e-10
Fuel Oil #4 Prod.	Barium	secondary	5.08e-09	2.57e-10
Fuel Oil #2 Prod.	Methyl chloride	secondary	5.05e-09	2.55e-10
Japanese Electric Grid	Methyl tert-butyl ether	model/secondary	4.75e-09	2.40e-10
Natural Gas Prod.	o-xylene	secondary	4.60e-09	2.32e-10
US electric grid	Phenol	model/secondary	4.54e-09	2.30e-10
LPG Production	Cumene	secondary	4.54e-09	2.29e-10
Japanese Electric Grid	Chromium (III)	model/secondary	4.48e-09	2.26e-10
LPG Production	Benzo[a]pyrene	secondary	4.28e-09	2.16e-10
US electric grid	Vinyl acetate	model/secondary	4.25e-09	2.15e-10
Fuel Oil #6 Prod.	Toluene	secondary	4.19e-09	2.12e-10
US electric grid	Phenanthrene	model/secondary	4.00e-09	2.02e-10
US electric grid	Ethylene dibromide	model/secondary	3.94e-09	1.99e-10
Japanese Electric Grid	1,1,1-Trichloroethane	model/secondary	3.66e-09	1.85e-10
Fuel Oil #2 Prod.	Di(2-ethylhexyl)phthalate	secondary	3.64e-09	1.84e-10
Japanese Electric Grid	Phenol	model/secondary	3.63e-09	1.83e-10
Fuel Oil #4 Prod.	Barium cmpds	secondary	3.62e-09	1.83e-10
US electric grid	Xylene (mixed isomers)	model/secondary	3.52e-09	1.78e-10
Japanese Electric Grid	Vinyl acetate	model/secondary	3.38e-09	1.71e-10
Natural Gas Prod.	Acrolein	secondary	3.37e-09	1.70e-10
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	3.26e-09	1.65e-10
US electric grid	Chromium (III)	model/secondary	3.15e-09	1.59e-10
Japanese Electric Grid	Ethylene dibromide	model/secondary	3.14e-09	1.59e-10
Fuel Oil #4 Prod.	Cadmium	secondary	3.04e-09	1.53e-10
Fuel Oil #2 Prod.	Acetaldehyde	secondary	2.90e-09	1.46e-10
Japanese Electric Grid	Xylene (mixed isomers)	model/secondary	2.81e-09	1.42e-10
Natural Gas Prod.	Methyl chloride	secondary	2.79e-09	1.41e-10
Fuel Oil #2 Prod.	Mercury	secondary	2.72e-09	1.37e-10

 Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
LPG Production	Biphenyl	secondary	2.71e-09	1.37e-10
Fuel Oil #6 Prod.	Isophorone	secondary	2.50e-09	1.27e-10
Fuel Oil #6 Prod.	Cadmium cmpds	secondary	2.35e-09	1.19e-10
LPG Production	Lead cmpds	secondary	2.33e-09	1.18e-10
Fuel Oil #4 Prod.	Carbon disulfide	secondary	2.29e-09	1.16e-10
LPG Production	Dibenzo[a,h]anthracene	secondary	2.23e-09	1.12e-10
LPG Production	Acenaphthylene	secondary	2.14e-09	1.08e-10
LPG Production	Anthracene	secondary	2.11e-09	1.07e-10
Fuel Oil #6 Prod.	1,4-Dichlorobenzene	secondary	2.07e-09	1.05e-10
Fuel Oil #2 Prod.	Dimethyl sulfate	secondary	2.02e-09	1.02e-10
Natural Gas Prod.	Di(2-ethylhexyl)phthalate	secondary	2.01e-09	1.02e-10
Fuel Oil #6 Prod.	Methyl ethyl ketone	secondary	2.00e-09	1.01e-10
Fuel Oil #6 Prod.	Tetrachloroethylene	secondary	1.98e-09	1.00e-10
LPG Production	Nickel cmpds	secondary	1.97e-09	9.96e-11
Fuel Oil #2 Prod.	Cadmium cmpds	secondary	1.95e-09	9.85e-11
LPG Production	Benzo[a]anthracene	secondary	1.91e-09	9.63e-11
Fuel Oil #2 Prod.	Toluene	secondary	1.90e-09	9.58e-11
Natural Gas Prod.	Mercury	secondary	1.88e-09	9.51e-11
US electric grid	2,3,7,8-TCDF	model/secondary	1.86e-09	9.41e-11
Japanese Electric Grid	Dibenzo[a,h]anthracene	model/secondary	1.86e-09	9.37e-11
Glass/frit	Nickel	primary	1.80e-09	9.11e-11
Fuel Oil #6 Prod.	1,2-Dichloroethane	secondary	1.76e-09	8.90e-11
Fuel Oil #4 Prod.	Benzyl chloride	secondary	1.75e-09	8.86e-11
US electric grid	1,1,1-Trichloroethane	model/secondary	1.73e-09	8.74e-11
Fuel Oil #4 Prod.	Cobalt	secondary	1.72e-09	8.71e-11
Fuel Oil #6 Prod.	Methyl methacrylate	secondary	1.71e-09	8.66e-11
Natural Gas Prod.	Acetaldehyde	secondary	1.60e-09	8.09e-11
Fuel Oil #6 Prod.	Styrene	secondary	1.51e-09	7.65e-11
Fuel Oil #4 Prod.	Aluminum (elemental)	secondary	1.50e-09	7.56e-11
Japanese Electric Grid	2,3,7,8-TCDF	model/secondary	1.49e-09	7.53e-11
LPG Production	Benzo[b,j,k]fluoranthene	secondary	1.47e-09	7.44e-11
LPG Production	Indeno(1,2,3-cd)pyrene	secondary	1.42e-09	7.16e-11
Fuel Oil #6 Prod.	Bromoform	secondary	1.41e-09	7.14e-11
Fuel Oil #6 Prod.	Dichloromethane	secondary	1.40e-09	7.05e-11
Japanese Electric Grid	o-xylene	model/secondary	1.28e-09	6.48e-11
Fuel Oil #4 Prod.	Chloroform	secondary	1.24e-09	6.25e-11
Japanese Electric Grid	Benzo[a]anthracene	model/secondary	1.22e-09	6.17e-11
US electric grid	Acenaphthene	model/secondary	1.21e-09	6.14e-11
LPG Production	Chrysene	secondary	1.20e-09	6.08e-11
US electric grid	Ethyl Chloride	model/secondary	1.15e-09	5.80e-11
Fuel Oil #4 Prod.	Propionaldehyde	secondary	1.14e-09	5.77e-11
Fuel Oil #6 Prod.	Chlorobenzene	secondary	1.13e-09	5.71e-11
Natural Gas Prod.	Dimethyl sulfate	secondary	1.11e-09	5.63e-11
Fuel Oil #2 Prod.	1,4-Dichlorobenzene	secondary	1.08e-09	5.44e-11
Natural Gas Prod.	Zinc (+2)	secondary	1.05e-09	5.30e-11
LPG Production	HALON-1301	secondary	1.02e-09	5.13e-11
Fuel Oil #2 Prod.	Isophorone	secondary	9.74e-10	4.92e-11
US electric grid	Cumene	model/secondary	9.71e-10	4.90e-11
Fuel Oil #6 Prod.	Aromatic hydrocarbons	secondary	9.47e-10	4.79e-11
Natural Gas Prod.	Chlorine	secondary	9.46e-10	4.78e-11

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Glass/frit	Chromium	primary	9.33e-10	4.72e-11
Japanese Electric Grid	Ethyl Chloride	model/secondary	9.16e-10	4.63e-11
Fuel Oil #6 Prod.	2,4-Dinitrotoluene	secondary	9.07e-10	4.58e-11
Fuel Oil #6 Prod.	Acetophenone	secondary	8.33e-10	4.21e-11
Fuel Oil #2 Prod.	Aromatic hydrocarbons	secondary	7.86e-10	3.97e-11
Fuel Oil #2 Prod.	Methyl ethyl ketone	secondary	7.79e-10	3.94e-11
Japanese Electric Grid	Benzo[b,j,k]fluoranthene	model/secondary	7.77e-10	3.93e-11
Fuel Oil #2 Prod.	Tetrachloroethylene	secondary	7.70e-10	3.89e-11
LPG Production	Benzo[b]fluoranthene	secondary	7.62e-10	3.85e-11
Natural Gas Prod.	Silicon	secondary	7.43e-10	3.75e-11
LPG Production	Mercury compounds	secondary	7.08e-10	3.58e-11
Fuel Oil #6 Prod.	Chromium (III)	secondary	6.96e-10	3.52e-11
Fuel Oil #2 Prod.	1,2-Dichloroethane	secondary	6.85e-10	3.46e-11
Japanese Electric Grid	Indeno(1,2,3-cd)pyrene	model/secondary	6.84e-10	3.45e-11
Fuel Oil #2 Prod.	Methyl methacrylate	secondary	6.66e-10	3.37e-11
LPG Production	Fluorene	secondary	6.60e-10	3.34e-11
Fuel Oil #4 Prod.	Acrolein	secondary	6.49e-10	3.28e-11
Glass/frit	Lead	primary	6.40e-10	3.23e-11
Fuel Oil #2 Prod.	Styrene	secondary	5.89e-10	2.97e-11
US electric grid	Biphenyl	model/secondary	5.79e-10	2.93e-11
LPG Production	Pyrene	secondary	5.71e-10	2.88e-11
LPG Production	Benzo[g,h,i]perylene	secondary	5.55e-10	2.80e-11
Fuel Oil #2 Prod.	Bromoform	secondary	5.50e-10	2.78e-11
Fuel Oil #2 Prod.	Dichloromethane	secondary	5.42e-10	2.74e-11
LPG Production	Fluoranthene	secondary	5.39e-10	2.72e-11
Fuel Oil #4 Prod.	Methyl chloride	secondary	5.39e-10	2.72e-11
Natural Gas Prod.	Isophorone	secondary	5.38e-10	2.72e-11
Japanese Electric Grid	Chrysene	model/secondary	5.38e-10	2.72e-11
Japanese Electric Grid	2-Methylnaphthalene	model/secondary	4.75e-10	2.40e-11
Glass/frit	Manganese	primary	4.67e-10	2.36e-11
Japanese Electric Grid	Biphenyl	model/secondary	4.62e-10	2.34e-11
Japanese Electric Grid	Anthracene	model/secondary	4.61e-10	2.33e-11
Fuel Oil #6 Prod.	Ethylbenzene	secondary	4.47e-10	2.26e-11
Fuel Oil #6 Prod.	Copper $(+1 \& +2)$	secondary	4.43e-10	2.24e-11
Fuel Oil #2 Prod.	Chlorobenzene	secondary	4.40e-10	2.22e-11
Natural Gas Prod.	Methyl ethyl ketone	secondary	4.31e-10	2.18e-11
Japanese Electric Grid	Benzo[g,h,i]perylene	model/secondary	4.30e-10	2.17e-11
Fuel Oil #6 Prod.	o-xylene	secondary	4.29e-10	2.17e-11
Natural Gas Prod.	Tetrachloroethylene	secondary	4.26e-10	2.15e-11
Fuel Oil #4 Prod.	Di(2-ethylhexyl)phthalate	secondary	3.88e-10	1.96e-11
Natural Gas Prod.	1.2-Dichloroethane	secondary	3.79e-10	1.91e-11
Natural Gas Prod.	Methyl methacrylate	secondary	3.68e-10	1.86e-11
Fuel Oil #2 Prod.	Copper $(+1 \& +2)$	secondary	3.68e-10	1.86e-11
US electric grid	Acenaphthylene	model/secondary	3.60e-10	1.82e-11
Fuel Oil #2 Prod.	2.4-Dinitrotoluene	secondary	3.52e-10	1.78e-11
US electric grid	Benzo[a]pyrene	model/secondary	3.41e-10	1.72e-11
US electric grid	Benzo[b.i.k]fluoranthene	model/secondary	3.40e-10	1.72e-11
Japanese Electric Grid	Acenaphthylene	model/secondary	3.29e-10	1.66e-11
Natural Gas Prod.	Styrene	secondary	3.25e-10	1.64e-11
Fuel Oil #2 Prod.	Acetophenone	secondary	3.24e-10	1.64e-11
		-		

 Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Barium cmpds	secondary	3.21e-10	1.62e-11
US electric grid	Anthracene	model/secondary	3.15e-10	1.59e-11
Fuel Oil #4 Prod.	Acetaldehyde	secondary	3.09e-10	1.56e-11
Natural Gas Prod.	Bromoform	secondary	3.04e-10	1.54e-11
Natural Gas Prod.	Dichloromethane	secondary	3.00e-10	1.52e-11
LPG Production	5-Methyl chrysene	secondary	2.94e-10	1.49e-11
Fuel Oil #4 Prod.	Mercury	secondary	2.89e-10	1.46e-11
Glass/frit	Cobalt	primary	2.87e-10	1.45e-11
Japanese Electric Grid	Benzo[a]pyrene	model/secondary	2.73e-10	1.38e-11
Fuel Oil #2 Prod.	Chromium (III)	secondary	2.70e-10	1.36e-11
Natural Gas Prod.	Chlorobenzene	secondary	2.43e-10	1.23e-11
LPG Production	Halogenated matter (organic)	secondary	2.33e-10	1.18e-11
US electric grid	Benzo[a]anthracene	model/secondary	2.26e-10	1.14e-11
Fuel Oil #6 Prod.	Methyl tert-butyl ether	secondary	2.25e-10	1.14e-11
Fuel Oil #4 Prod.	Dimethyl sulfate	secondary	2.15e-10	1.09e-11
Fuel Oil #2 Prod.	o-xvlene	secondary	2.06e-10	1.04e-11
Japanese Electric Grid	Pyrene	model/secondary	2.00e-10	1.01e-11
Natural Gas Prod.	2.4-Dinitrotoluene	secondary	1.95e-10	9.85e-12
Fuel Oil #4 Prod.	Toluene	secondary	1.90e-10	9.57e-12
Natural Gas Prod	Acetophenone	secondary	1 79e-10	9.04e-12
Fuel Oil #2 Prod.	Ethylbenzene	secondary	1.74e-10	8.80e-12
Japanese Electric Grid	Fluorene	model/secondary	1 74e-10	8 80e-12
Natural Gas Prod	3-Methylcholanthrene	secondary	1.73e-10	8 73e-12
Fuel Oil #6 Prod	Phenanthrene	secondary	1 71e-10	8.65e-12
Japanese Electric Grid	Fluoranthene	model/secondary	1.66e-10	8 40e-12
US electric grid	Chrysene	model/secondary	1.64e-10	8 28e-12
Natural Gas Prod	Chromium (III)	secondary	1.61e-10	8 14e-12
US electric grid	Indeno(1.2.3-cd)nyrene	model/secondary	1.61e-10	8 12e-12
Fuel Oil #6 Prod	Vinyl acetate	secondary	1.60e-10	8.09e-12
Fuel Oil #4 Prod	Cadmium cmpds	secondary	1 58e-10	8.00e-12
Fuel Oil #6 Prod	Ethylene dibromide	secondary	1 48e-10	7 50e-12
US electric grid	Fluorene	model/secondary	1 28e-10	6.45e-12
Fuel Oil #6 Prod	3-Methylcholanthrene	secondary	1 10e-10	5 54e-12
Natural Gas Prod	2-Methylnaphthalene	secondary	1.08e-10	5.47e-12
Fuel Oil #6 Prod	$Z \operatorname{inc}(+2)$	secondary	1.07e-10	5 38e-12
Fuel Oil #4 Prod	Isophorone	secondary	1.04e-10	5.25e-12
US electric grid	Fluoranthene	model/secondary	1.07e-10	5.25e-12
Fuel Oil #4 Prod	1 4-Dichlorobenzene	secondary	1.02e-10	5.15e-12
Natural Gas Prod	Ethylhenzene	secondary	9.66e-11	4.88e-12
US electric grid	Dibenzo[a h]anthracene	model/secondary	8 75e-11	4.00C-12
Fuel Oil #2 Prod	Methyl tert butyl ether	secondary	8.73e-11 8.74e-11	4.42e-12
US alastria grid	2 Methylnenhthelene	model/secondem/	8.740-11	4.420-12
Natural Gas Brad	2-Methymaphinalene	secondary	8.510-11	4.30e-12
US electric grid	Purene	model/secondary	8.30 C- 11	4.290-12
Eval Oil #4 Prod	r yiciic Mathyi athyi katana	model/secondary	8 21o 11	4.20e-12
Fuel Oil #4 Prod	Totrochlorocthylono	secondary	8.210-11	4.206-12
Fuel Oil #4 Prod	1.2 Dichloroethane	secondary	0.21e-11 7 20c 11	4.130-12
Fuel Oil #4 Prod	1,2-Dichioloculate	secondary	7.10-11	2.50x 12
Natural Cas Brad	Dhamanthrana	secondary	7.100-11	5.59e-12
Eval Oil #2 Prod	Phononthrono	secondary	7.026-11	3.50e-12 2.55a 12
Fuel Oil #2 Prou.	2 Mothylpophtholog	secondary	/.03e-11	5.55e-12
ruei Oli #0 Prou.	2-ivieuryinaphtnaiene	secondary	6.8/e-11	3.4/e-12

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Fuel Oil #2 Prod.	Zinc (+2)	secondary	6.78e-11	3.42e-12
Fuel Oil #4 Prod.	Aromatic hydrocarbons	secondary	6.39e-11	3.23e-12
US electric grid	5-Methyl chrysene	model/secondary	6.30e-11	3.18e-12
Fuel Oil #4 Prod.	Styrene	secondary	6.28e-11	3.17e-12
Fuel Oil #2 Prod.	Vinyl acetate	secondary	6.23e-11	3.15e-12
Fuel Oil #6 Prod.	1,1,1-Trichloroethane	secondary	6.11e-11	3.08e-12
US electric grid	o-xylene	model/secondary	6.06e-11	3.06e-12
Fuel Oil #6 Prod.	Chlorine	secondary	5.92e-11	2.99e-12
Fuel Oil #4 Prod.	Bromoform	secondary	5.86e-11	2.96e-12
Fuel Oil #4 Prod.	Dichloromethane	secondary	5.78e-11	2.92e-12
Fuel Oil #2 Prod.	Ethylene dibromide	secondary	5.77e-11	2.92e-12
US electric grid	Benzo[g,h,i]perylene	model/secondary	5.75e-11	2.91e-12
Fuel Oil #2 Prod.	3-Methylcholanthrene	secondary	5.70e-11	2.88e-12
Japanese Electric Grid	5-Methyl chrysene	model/secondary	5.03e-11	2.54e-12
Natural Gas Prod.	Methyl tert-butyl ether	secondary	4.83e-11	2.44e-12
Fuel Oil #4 Prod.	Chlorobenzene	secondary	4.69e-11	2.37e-12
Fuel Oil #6 Prod.	Acenaphthene	secondary	4.67e-11	2.36e-12
Fuel Oil #6 Prod.	Ethyl Chloride	secondary	4.33e-11	2.19e-12
Fuel Oil #4 Prod.	2.4-Dinitrotoluene	secondary	3.76e-11	1.90e-12
Fuel Oil #6 Prod.	Cumene	secondary	3.66e-11	1.85e-12
Fuel Oil #2 Prod.	2-Methylnaphthalene	secondary	3.57e-11	1.80e-12
Fuel Oil #4 Prod.	Acetophenone	secondary	3.45e-11	1.74e-12
Natural Gas Prod.	Vinvl acetate	secondary	3.44e-11	1.74e-12
Fuel Oil #6 Prod.	Benzo[a]pyrene	secondary	3.20e-11	1.62e-12
Natural Gas Prod.	Ethylene dibromide	secondary	3.19e-11	1.61e-12
Fuel Oil #2 Prod.	Chlorine	secondary	3.12e-11	1.58e-12
Fuel Oil #4 Prod.	Copper $(+1 \& +2)$	secondary	2.99e-11	1.51e-12
Fuel Oil #4 Prod.	Chromium (III)	secondary	2.88e-11	1.45e-12
Fuel Oil #2 Prod.	1.1.1-Trichloroethane	secondary	2.37e-11	1.20e-12
Fuel Oil #6 Prod.	Biphenyl	secondary	2.18e-11	1.10e-12
Fuel Oil #2 Prod.	Acenaphthene	secondary	2.13e-11	1.08e-12
Fuel Oil #4 Prod.	o-xvlene	secondary	2.01e-11	1.02e-12
Natural Gas Prod.	Benzo[a]pyrene	secondary	1.97e-11	9.97e-13
Fuel Oil #4 Prod.	Ethylbenzene	secondary	1.86e-11	9.37e-13
Natural Gas Prod.	Dibenzo[a.h]anthracene	secondary	1.78e-11	8.98e-13
Natural Gas Prod.	Acenaphthene	secondary	1.77e-11	8.96e-13
Fuel Oil #2 Prod.	Ethyl Chloride	secondary	1.68e-11	8.50e-13
Fuel Oil #6 Prod	Acenaphthylene	secondary	1.61e-11	8 15e-13
Fuel Oil #6 Prod.	Anthracene	secondary	1.52e-11	7.69e-13
Fuel Oil #2 Prod	Cumene	secondary	1 42e-11	7 19e-13
Natural Gas Prod	Cadmium empds	secondary	1 40e-11	7.09e-13
Fuel Oil #2 Prod	Benzo[a]pyrene	secondary	1 33e-11	6 70e-13
Natural Gas Prod	1 1 1-Trichloroethane	secondary	1.35e-11	6.63e-13
Fuel Oil #6 Prod	Dibenzo[a h]anthracene	secondary	1.9e-11	6.03e-13
Fuel Oil #6 Prod	Benzo[a]anthracene	secondary	1.19e-11	6.02e-13
Fuel Oil #6 Prod	Benzo[h i k]fluoranthene	secondary	1 19e-11	6.00e-13
Fuel Oil #6 Prod	Indeno(1.2.3-cd)nvrene	secondary	9 33e-12	4 71e-13
Fuel Oil #4 Prod	Methyl tert-hutyl ether	secondary	0 320 12	4.716-13
Natural Gas Prod	Ethyl Chloride	secondary	9.32c-12	4 70e-13
Natural Gas Prod	Benzo[a]anthracana	secondary	8 560 12	4 320 12
Tutural Oas 1100.	Denzolajantinacene	Secondary	0.500-12	T.J2C-13

 Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Fuel Oil #2 Prod.	Biphenyl	secondary	8.49e-12	4.29e-13
Fuel Oil #6 Prod.	Chrysene	secondary	8.28e-12	4.19e-13
Natural Gas Prod.	Anthracene	secondary	8.13e-12	4.11e-13
Fuel Oil #6 Prod.	Lead cmpds	secondary	7.89e-12	3.99e-13
Natural Gas Prod.	Cumene	secondary	7.87e-12	3.98e-13
Natural Gas Prod.	Indeno(1,2,3-cd)pyrene	secondary	7.53e-12	3.80e-13
Fuel Oil #4 Prod.	Phenanthrene	secondary	7.32e-12	3.70e-13
Natural Gas Prod.	Acenaphthylene	secondary	7.00e-12	3.54e-13
Fuel Oil #6 Prod.	Nickel cmpds	secondary	6.67e-12	3.37e-13
Fuel Oil #4 Prod.	Vinyl acetate	secondary	6.64e-12	3.35e-13
Fuel Oil #2 Prod.	Acenaphthylene	secondary	6.63e-12	3.35e-13
Fuel Oil #2 Prod.	Dibenzo[a,h]anthracene	secondary	6.56e-12	3.31e-13
Fuel Oil #2 Prod.	Lead cmpds	secondary	6.55e-12	3.31e-13
Fuel Oil #2 Prod.	Anthracene	secondary	6.49e-12	3.28e-13
Natural Gas Prod.	Benzo[b]fluoranthene	secondary	6.37e-12	3.22e-13
Fuel Oil #4 Prod.	Ethylene dibromide	secondary	6.16e-12	3.11e-13
Fuel Oil #4 Prod.	Zinc (+2)	secondary	5.97e-12	3.02e-13
Fuel Oil #2 Prod.	Benzo[a]anthracene	secondary	5.73e-12	2.90e-13
Natural Gas Prod.	Aromatic hydrocarbons	secondary	5.66e-12	2.86e-13
Fuel Oil #2 Prod.	Nickel cmpds	secondary	5.53e-12	2.80e-13
Natural Gas Prod.	Chrysene	secondary	5.43e-12	2.74e-13
Fuel Oil #4 Prod.	3-Methylcholanthrene	secondary	5.39e-12	2.73e-13
Fuel Oil #6 Prod.	Fluorene	secondary	5.11e-12	2.58e-13
Natural Gas Prod.	Biphenyl	secondary	4.70e-12	2.37e-13
Fuel Oil #2 Prod.	Benzo[b,j,k]fluoranthene	secondary	4.62e-12	2.33e-13
Fuel Oil #2 Prod.	Indeno(1,2,3-cd)pyrene	secondary	4.30e-12	2.17e-13
Fuel Oil #6 Prod.	Benzo[b]fluoranthene	secondary	4.19e-12	2.11e-13
Fuel Oil #6 Prod.	Fluoranthene	secondary	4.11e-12	2.08e-13
Fuel Oil #6 Prod.	Pyrene	secondary	4.04e-12	2.04e-13
Fuel Oil #2 Prod.	Chrysene	secondary	3.68e-12	1.86e-13
Fuel Oil #6 Prod.	HALON-1301	secondary	3.43e-12	1.74e-13
Fuel Oil #6 Prod.	Benzo[g,h,i]perylene	secondary	3.43e-12	1.73e-13
Fuel Oil #4 Prod.	2-Methylnaphthalene	secondary	3.38e-12	1.71e-13
Natural Gas Prod.	Benzo[g,h,i]perylene	secondary	3.19e-12	1.61e-13
Fuel Oil #4 Prod.	Chlorine	secondary	2.94e-12	1.48e-13
Fuel Oil #2 Prod.	HALON-1301	secondary	2.85e-12	1.44e-13
Natural Gas Prod.	Copper (+1 & +2)	secondary	2.65e-12	1.34e-13
Natural Gas Prod.	Benzo[b,j,k]fluoranthene	secondary	2.55e-12	1.29e-13
Fuel Oil #4 Prod.	1,1,1-Trichloroethane	secondary	2.53e-12	1.28e-13
Natural Gas Prod.	Pyrene	secondary	2.45e-12	1.24e-13
Fuel Oil #6 Prod.	Mercury compounds	secondary	2.39e-12	1.21e-13
Fuel Oil #6 Prod.	5-Methyl chrysene	secondary	2.38e-12	1.20e-13
Fuel Oil #2 Prod.	Benzo[b]fluoranthene	secondary	2.25e-12	1.14e-13
Fuel Oil #4 Prod.	Acenaphthene	secondary	2.12e-12	1.07e-13
Fuel Oil #2 Prod.	Fluorene	secondary	2.06e-12	1.04e-13
Fuel Oil #2 Prod.	Mercury compounds	secondary	1.99e-12	1.00e-13
Fuel Oil #4 Prod.	Ethyl Chloride	secondary	1.79e-12	9.06e-14
Fuel Oil #2 Prod.	Pyrene	secondary	1.75e-12	8.85e-14
Fuel Oil #2 Prod.	Fluoranthene	secondary	1.67e-12	8.46e-14
Fuel Oil #2 Prod.	Benzo[g,h,i]perylene	secondary	1.67e-12	8.42e-14

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Fluorene	secondary	1.64e-12	8.28e-14
Fuel Oil #4 Prod.	Cumene	secondary	1.52e-12	7.67e-14
Natural Gas Prod.	Fluoranthene	secondary	1.46e-12	7.40e-14
Fuel Oil #4 Prod.	Benzo[a]pyrene	secondary	1.37e-12	6.95e-14
Fuel Oil #2 Prod.	5-Methyl chrysene	secondary	9.24e-13	4.67e-14
Fuel Oil #4 Prod.	Biphenyl	secondary	9.06e-13	4.58e-14
Fuel Oil #6 Prod.	Halogenated matter (organic)	secondary	7.89e-13	3.99e-14
Fuel Oil #4 Prod.	Acenaphthylene	secondary	6.90e-13	3.49e-14
Fuel Oil #4 Prod.	Anthracene	secondary	6.65e-13	3.36e-14
Fuel Oil #2 Prod.	Halogenated matter (organic)	secondary	6.55e-13	3.31e-14
Fuel Oil #4 Prod.	Dibenzo[a,h]anthracene	secondary	6.08e-13	3.07e-14
LPG Production	Halogenated hydrocarbons (unspecified)	secondary	5.84e-13	2.95e-14
Fuel Oil #4 Prod.	Benzo[a]anthracene	secondary	5.59e-13	2.82e-14
Fuel Oil #4 Prod.	Lead cmpds	secondary	5.32e-13	2.69e-14
Natural Gas Prod.	5-Methyl chrysene	secondary	5.11e-13	2.58e-14
Fuel Oil #4 Prod.	Benzo[b,j,k]fluoranthene	secondary	4.92e-13	2.49e-14
Fuel Oil #4 Prod.	Nickel cmpds	secondary	4.50e-13	2.27e-14
Fuel Oil #4 Prod.	Indeno(1,2,3-cd)pyrene	secondary	4.26e-13	2.15e-14
Fuel Oil #4 Prod.	Chrysene	secondary	3.70e-13	1.87e-14
Fuel Oil #4 Prod.	HALON-1301	secondary	2.32e-13	1.17e-14
Fuel Oil #4 Prod.	Fluorene	secondary	2.16e-13	1.09e-14
Fuel Oil #4 Prod.	Benzo[b]fluoranthene	secondary	2.10e-13	1.06e-14
Fuel Oil #4 Prod.	Pyrene	secondary	1.78e-13	9.00e-15
Fuel Oil #4 Prod.	Fluoranthene	secondary	1.75e-13	8.83e-15
Fuel Oil #4 Prod.	Benzo[g,h,i]perylene	secondary	1.62e-13	8.18e-15
Fuel Oil #4 Prod.	Mercury compounds	secondary	1.61e-13	8.16e-15
Fuel Oil #4 Prod.	5-Methyl chrysene	secondary	9.85e-14	4.98e-15
Fuel Oil #4 Prod.	Halogenated matter (organic)	secondary	5.32e-14	2.69e-15
Natural Gas Prod.	Lead cmpds	secondary	4.72e-14	2.38e-15
Natural Gas Prod.	Nickel cmpds	secondary	3.99e-14	2.01e-15
Natural Gas Prod.	HALON-1301	secondary	2.05e-14	1.04e-15
Natural Gas Prod.	Mercury compounds	secondary	1.43e-14	7.23e-16
Natural Gas Prod.	Halogenated matter (organic)	secondary	4.72e-15	2.38e-16
Fuel Oil #6 Prod.	Halogenated hydrocarbons (unspecified)	secondary	1.97e-15	9.98e-17
Fuel Oil #2 Prod.	Halogenated hydrocarbons (unspecified)	secondary	1.64e-15	8.27e-17
Fuel Oil #4 Prod.	Halogenated hydrocarbons (unspecified)	secondary	1.33e-16	6.72e-18
Natural Gas Prod.	Halogenated hydrocarbons (unspecified)	secondary	1.18e-17	5.95e-19
Total Manufacturing			9.77e+01	4.94e+00
Use. Maintenance and Repair I	ife-cycle Stage			
US electric grid	Sulfur dioxide	model/secondary	1 65e+03	8 32e+01
US electric grid	Nitrogen oxides	model/secondary	2 35e+00	1 19e-01
US electric grid	Methane	model/secondary	1.29e+00	6 52e-02
US electric grid	Carbon monoxide	model/secondary	1.09e+00	5.51e-02
US electric grid	Arsenic	model/secondary	5 58e-01	2.82e-02
US electric grid	Hydrochloric acid	model/secondary	4 93e-01	2.020 02 2.49e-02
US electric grid	PM-10	model/secondary	1.16e-01	5.84e-03
US electric grid	Selenium	model/secondary	9 28e-02	4.69e-03
US electric grid	Vanadium	model/secondary	6.98e-02	3.53e-03
US electric grid	Hydrofluoric acid	model/secondary	2.69e-02	1.36e-03
US electric grid	Formaldehyde	model/secondary	9 70e-03	4.90e-04
		,		

 Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
US electric grid	Benzene	model/secondary	6.98e-03	3.53e-04
US electric grid	Nitrous oxide	model/secondary	6.80e-03	3.43e-04
US electric grid	Phosphorus (yellow or white)	model/secondary	3.93e-03	1.99e-04
US electric grid	Zinc (elemental)	model/secondary	1.81e-03	9.16e-05
US electric grid	Antimony	model/secondary	1.49e-03	7.52e-05
US electric grid	Chromium (VI)	model/secondary	5.37e-04	2.71e-05
US electric grid	Molybdenum	model/secondary	4.97e-04	2.51e-05
US electric grid	Methyl hydrazine	model/secondary	4.61e-04	2.33e-05
US electric grid	2-Chloroacetophenone	model/secondary	4.32e-04	2.18e-05
US electric grid	Bromomethane	model/secondary	4.27e-04	2.16e-05
US electric grid	Nickel	model/secondary	2.32e-04	1.17e-05
US electric grid	Cyanide (-1)	model/secondary	2.24e-04	1.13e-05
US electric grid	2,3,7,8-TCDD	model/secondary	1.70e-04	8.58e-06
US electric grid	Naphthalene	model/secondary	1.37e-04	6.92e-06
US electric grid	Barium	model/secondary	1.16e-04	5.88e-06
US electric grid	Carbon disulfide	model/secondary	9.18e-05	4.64e-06
US electric grid	Benzyl chloride	model/secondary	7.02e-05	3.55e-06
US electric grid	Cadmium	model/secondary	5.48e-05	2.77e-06
US electric grid	Chloroform	model/secondary	4.95e-05	2.50e-06
US electric grid	Propionaldehyde	model/secondary	4.58e-05	2.31e-06
US electric grid	Manganese	model/secondary	4.56e-05	2.30e-06
US electric grid	Fluoride	model/secondary	3.91e-05	1.97e-06
US electric grid	Beryllium	model/secondary	2.92e-05	1.47e-06
US electric grid	Acrolein	model/secondary	2.60e-05	1.31e-06
US electric grid	Lead	model/secondary	2.55e-05	1.29e-06
US electric grid	Cobalt	model/secondary	2.43e-05	1.23e-06
US electric grid	Copper	model/secondary	2.24e-05	1.13e-06
US electric grid	Methyl chloride	model/secondary	2.16e-05	1.09e-06
US electric grid	Di(2-ethylhexyl)phthalate	model/secondary	1.55e-05	7.85e-07
US electric grid	Acetaldehyde	model/secondary	1.24e-05	6.25e-07
US electric grid	Hexane	model/secondary	1.20e-05	6.07e-07
US electric grid	Dimethyl sulfate	model/secondary	8.61e-06	4.35e-07
US electric grid	Mercury	model/secondary	7.51e-06	3.80e-07
US electric grid	Toluene	model/secondary	4.25e-06	2.15e-07
US electric grid	Isophorone	model/secondary	4.16e-06	2.10e-07
US electric grid	Methyl ethyl ketone	model/secondary	3.33e-06	1.68e-07
US electric grid	Tetrachloroethylene	model/secondary	3.29e-06	1.66e-07
US electric grid	1,2-Dichloroethane	model/secondary	2.93e-06	1.48e-07
US electric grid	Methyl methacrylate	model/secondary	2.85e-06	1.44e-07
US electric grid	Styrene	model/secondary	2.51e-06	1.27e-07
US electric grid	Bromoform	model/secondary	2.35e-06	1.19e-07
US electric grid	Dichloromethane	model/secondary	2.32e-06	1.17e-07
US electric grid	Chlorobenzene	model/secondary	1.88e-06	9.49e-08
US electric grid	2,4-Dinitrotoluene	model/secondary	1.51e-06	7.61e-08
US electric grid	Acetophenone	model/secondary	1.38e-06	6.99e-08
US electric grid	Ethylbenzene	model/secondary	7.41e-07	3.74e-08
US electric grid	Methyl tert-butyl ether	model/secondary	3.74e-07	1.89e-08
US electric grid	Phenol	model/secondary	2.85e-07	1.44e-08
US electric grid	Vinyl acetate	model/secondary	2.66e-07	1.34e-08
US electric grid	Phenanthrene	model/secondary	2.51e-07	1.27e-08

 Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
US electric grid	Ethylene dibromide	model/secondary	2.47e-07	1.25e-08
US electric grid	Xylene (mixed isomers)	model/secondary	2.21e-07	1.11e-08
US electric grid	Chromium (III)	model/secondary	1.97e-07	9.96e-09
US electric grid	2,3,7,8-TCDF	model/secondary	1.17e-07	5.89e-09
US electric grid	1,1,1-Trichloroethane	model/secondary	1.08e-07	5.48e-09
US electric grid	Acenaphthene	model/secondary	7.61e-08	3.84e-09
US electric grid	Ethyl Chloride	model/secondary	7.19e-08	3.63e-09
US electric grid	Cumene	model/secondary	6.08e-08	3.07e-09
US electric grid	Biphenyl	model/secondary	3.63e-08	1.83e-09
US electric grid	Acenaphthylene	model/secondary	2.26e-08	1.14e-09
US electric grid	Benzo[a]pyrene	model/secondary	2.14e-08	1.08e-09
US electric grid	Benzo[b,j,k]fluoranthene	model/secondary	2.13e-08	1.08e-09
US electric grid	Anthracene	model/secondary	1.97e-08	9.96e-10
US electric grid	Benzo[a]anthracene	model/secondary	1.42e-08	7.15e-10
US electric grid	Chrysene	model/secondary	1.03e-08	5.19e-10
US electric grid	Indeno(1,2,3-cd)pyrene	model/secondary	1.01e-08	5.08e-10
US electric grid	Fluorene	model/secondary	7.99e-09	4.04e-10
US electric grid	Fluoranthene	model/secondary	6.39e-09	3.23e-10
US electric grid	Dibenzo[a,h]anthracene	model/secondary	5.48e-09	2.77e-10
US electric grid	2-Methylnaphthalene	model/secondary	5.33e-09	2.69e-10
US electric grid	Pyrene	model/secondary	5.28e-09	2.67e-10
US electric grid	5-Methyl chrysene	model/secondary	3.95e-09	1.99e-10
US electric grid	o-xylene	model/secondary	3.79e-09	1.92e-10
US electric grid	Benzo[g,h,i]perylene	model/secondary	3.60e-09	1.82e-10
Total Use, Maintenance and Re	pair		1.65e+03	8.35e+01
End-of-life Life-cycle Stage				
CRT Incineration	Sulfur dioxide	secondary	1.98e-01	1.00e-02
CRT landfilling	Sulfur dioxide	primary	1.85e-01	9.34e-03
US electric grid	Sulfur dioxide	model/secondary	1.65e-01	8.32e-03
CRT landfilling	Carbon monoxide	primary	9.95e-03	5.03e-04
CRT landfilling	Nitrogen dioxide	primary	3.71e-03	1.87e-04
CRT landfilling	PM	primary	4.17e-04	2.11e-05
US electric grid	Nitrogen oxides	model/secondary	2.36e-04	1.19e-05
US electric grid	Methane	model/secondary	1.29e-04	6.52e-06
US electric grid	Carbon monoxide	model/secondary	1.09e-04	5.52e-06
CRT landfilling	Methane	primary	1.09e-04	5.50e-06
CRT landfilling	Arsenic cmpds	primary	1.02e-04	5.17e-06
CRT Incineration	Arsenic cmpds	secondary	9.83e-05	4.97e-06
US electric grid	Arsenic	model/secondary	5.58e-05	2.82e-06
LPG Production	Carbon monoxide	secondary	4.98e-05	2.52e-06
US electric grid	Hydrochloric acid	model/secondary	4.93e-05	2.49e-06
CRT landfilling	Benzene	primary	4.05e-05	2.04e-06
CRT Incineration	Lead	secondary	2.86e-05	1.44e-06
LPG Production	Methane	secondary	1.45e-05	7.32e-07
LPG Production	Sulfur oxides	secondary	1.38e-05	6.99e-07
CRT Incineration	Barium cmpds	secondary	1.27e-05	6.43e-07
US electric grid	PM-10	model/secondary	1.16e-05	5.84e-07
CRT Incineration	Silver compounds	secondary	1.12e-05	5.67e-07
CRT landfilling	Silver compounds	primary	1.05e-05	5.29e-07
CRT landfilling	Barium cmpds	primary	1.01e-05	5.12e-07

 Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
LPG Production	Nitrogen oxides	secondary	9.89e-06	5.00e-07
US electric grid	Selenium	model/secondary	9.28e-06	4.69e-07
LPG Production	Vanadium	secondary	9.04e-06	4.57e-07
LPG Production	Benzene	secondary	7.41e-06	3.74e-07
US electric grid	Vanadium	model/secondary	6.99e-06	3.53e-07
CRT landfilling	Hydrochloric acid	primary	4.69e-06	2.37e-07
CRT landfilling	Ammonia	primary	3.81e-06	1.92e-07
US electric grid	Hydrofluoric acid	model/secondary	2.69e-06	1.36e-07
LPG Production	PM	secondary	2.23e-06	1.12e-07
LPG Production	Arsenic	secondary	1.78e-06	9.01e-08
LPG Production	Formaldehyde	secondary	1.16e-06	5.85e-08
CRT landfilling	Cadmium cmpds	primary	9.72e-07	4.91e-08
US electric grid	Formaldehyde	model/secondary	9.70e-07	4.90e-08
CRT Incineration	Cadmium cmpds	secondary	9.41e-07	4.76e-08
US electric grid	Benzene	model/secondary	6.98e-07	3.53e-08
US electric grid	Nitrous oxide	model/secondary	6.80e-07	3.44e-08
CRT landfilling	Hydrogen sulfide	primary	4.61e-07	2.33e-08
CRT landfilling	Chromium (VI)	primary	4.44e-07	2.24e-08
US electric grid	Phosphorus (yellow or white)	model/secondary	3.93e-07	1.99e-08
LPG Production	Hydrochloric acid	secondary	3.18e-07	1.60e-08
LPG Production	Nitrous oxide	secondary	2.80e-07	1.41e-08
US electric grid	Zinc (elemental)	model/secondary	1.81e-07	9.16e-09
US electric grid	Antimony	model/secondary	1.49e-07	7.52e-09
LPG Production	Phosphorus (yellow or white)	secondary	8.47e-08	4.28e-09
LPG Production	Fluorides (F-)	secondary	6.51e-08	3.29e-09
LPG Production	Selenium	secondary	6.40e-08	3.23e-09
US electric grid	Chromium (VI)	model/secondary	5.37e-08	2.72e-09
LPG Production	Hydrogen sulfide	secondary	5.29e-08	2.67e-09
LPG Production	Ammonia	secondary	5.11e-08	2.58e-09
US electric grid	Molybdenum	model/secondary	4.97e-08	2.51e-09
US electric grid	Methyl hydrazine	model/secondary	4.61e-08	2.33e-09
US electric grid	2-Chloroacetophenone	model/secondary	4.32e-08	2.18e-09
US electric grid	Bromomethane	model/secondary	4.27e-08	2.16e-09
CRT landfilling	Selenium	primary	3.21e-08	1.62e-09
US electric grid	Nickel	model/secondary	2.32e-08	1.17e-09
US electric grid	Cyanide (-1)	model/secondary	2.24e-08	1.13e-09
CRT Incineration	Carbon tetrachloride	secondary	1.81e-08	9.16e-10
LPG Production	Hydrofluoric acid	secondary	1.73e-08	8.75e-10
US electric grid	2,3,7,8-TCDD	model/secondary	1.70e-08	8.59e-10
CRT landfilling	Carbon tetrachloride	primary	1.69e-08	8.54e-10
LPG Production	Molybdenum	secondary	1.42e-08	7.19e-10
US electric grid	Naphthalene	model/secondary	1.37e-08	6.92e-10
LPG Production	Chromium (VI)	secondary	1.36e-08	6.88e-10
LPG Production	Ethane	secondary	1.34e-08	6.77e-10
CRT landfilling	Chloroform	primary	1.32e-08	6.67e-10
US electric grid	Barium	model/secondary	1.17e-08	5.89e-10
CRT landfilling	Mercury compounds	primary	1.15e-08	5.84e-10
CRT Incineration	Mercury compounds	secondary	1.13e-08	5.72e-10
LPG Production	Zinc (elemental)	secondary	1.04e-08	5.23e-10
US electric grid	Carbon disulfide	model/secondary	9.18e-09	4.64e-10

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
CRT landfilling	Toluene	primary	8.22e-09	4.15e-10
LPG Production	Hexane	secondary	7.78e-09	3.93e-10
US electric grid	Benzyl chloride	model/secondary	7.03e-09	3.55e-10
LPG Production	Phenol	secondary	6.17e-09	3.12e-10
LPG Production	Pentane	secondary	5.62e-09	2.84e-10
US electric grid	Cadmium	model/secondary	5.48e-09	2.77e-10
US electric grid	Chloroform	model/secondary	4.96e-09	2.50e-10
US electric grid	Propionaldehyde	model/secondary	4.58e-09	2.31e-10
US electric grid	Manganese	model/secondary	4.56e-09	2.30e-10
US electric grid	Fluoride	model/secondary	3.91e-09	1.97e-10
LPG Production	PM-10	secondary	3.88e-09	1.96e-10
LPG Production	Nickel	secondary	3.56e-09	1.80e-10
US electric grid	Beryllium	model/secondary	2.92e-09	1.47e-10
US electric grid	Acrolein	model/secondary	2.60e-09	1.31e-10
US electric grid	Lead	model/secondary	2.55e-09	1.29e-10
LPG Production	Aluminum (+3)	secondary	2.44e-09	1.24e-10
US electric grid	Cobalt	model/secondary	2.43e-09	1.23e-10
US electric grid	Copper	model/secondary	2.24e-09	1.13e-10
US electric grid	Methyl chloride	model/secondary	2.16e-09	1.09e-10
LPG Production	Antimony	secondary	1.85e-09	9.35e-11
CRT landfilling	Xylene (mixed isomers)	primary	1.69e-09	8.56e-11
CRT Incineration	Lead cmpds	secondary	1.62e-09	8.19e-11
CRT landfilling	Lead cmpds	primary	1.58e-09	7.98e-11
US electric grid	Di(2-ethylhexyl)phthalate	model/secondary	1.55e-09	7.86e-11
US electric grid	Acetaldehyde	model/secondary	1.24e-09	6.26e-11
CRT landfilling	Tetrachloroethylene	primary	1.20e-09	6.08e-11
US electric grid	Hexane	model/secondary	1.20e-09	6.07e-11
CRT landfilling	1,2-Dichloroethane	primary	1.15e-09	5.81e-11
LPG Production	Nitrate	secondary	1.05e-09	5.28e-11
US electric grid	Dimethyl sulfate	model/secondary	8.61e-10	4.35e-11
CRT Incineration	Trichloroethylene	secondary	7.65e-10	3.86e-11
US electric grid	Mercury	model/secondary	7.52e-10	3.80e-11
CRT landfilling	Trichloroethylene	primary	7.13e-10	3.60e-11
CRT landfilling	Ethylbenzene	primary	4.44e-10	2.24e-11
US electric grid	Toluene	model/secondary	4.25e-10	2.15e-11
US electric grid	Isophorone	model/secondary	4.16e-10	2.10e-11
US electric grid	Methyl ethyl ketone	model/secondary	3.33e-10	1.68e-11
US electric grid	Tetrachloroethylene	model/secondary	3.29e-10	1.66e-11
LPG Production	Copper	secondary	3.01e-10	1.52e-11
LPG Production	Methyl hydrazine	secondary	2.97e-10	1.50e-11
US electric grid	1.2-Dichloroethane	model/secondary	2.93e-10	1.48e-11
LPG Production	Silicon	secondary	2.92e-10	1.48e-11
US electric grid	Methyl methacrylate	model/secondary	2.85e-10	1.44e-11
LPG Production	2-Chloroacetophenone	secondary	2.78e-10	1.41e-11
LPG Production	Dimethylbenzanthracene	secondary	2.75e-10	1.39e-11
LPG Production	Bromomethane	secondary	2.75e-10	1.39e-11
US electric grid	Styrene	model/secondary	2.52e-10	1.27e-11
US electric grid	Bromoform	model/secondary	2.35e-10	1.19e-11
US electric grid	Dichloromethane	model/secondarv	2.32e-10	1.17e-11
LPG Production	Naphthalene	secondary	2.26e-10	1.14e-11
	1			

 Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
LPG Production	Lead	secondary	2.13e-10	1.08e-11
CRT landfilling	Dichloromethane	primary	2.01e-10	1.01e-11
LPG Production	Manganese	secondary	1.93e-10	9.73e-12
US electric grid	Chlorobenzene	model/secondary	1.88e-10	9.50e-12
LPG Production	Beryllium	secondary	1.85e-10	9.36e-12
CRT Incineration	Vinyl chloride	secondary	1.61e-10	8.14e-12
LPG Production	Barium	secondary	1.61e-10	8.13e-12
US electric grid	2,4-Dinitrotoluene	model/secondary	1.51e-10	7.61e-12
CRT landfilling	Vinyl chloride	primary	1.50e-10	7.59e-12
LPG Production	Cyanide (-1)	secondary	1.44e-10	7.30e-12
US electric grid	Acetophenone	model/secondary	1.38e-10	6.99e-12
LPG Production	Barium cmpds	secondary	1.37e-10	6.92e-12
LPG Production	Cadmium	secondary	7.95e-11	4.02e-12
US electric grid	Ethylbenzene	model/secondary	7.41e-11	3.74e-12
LPG Production	Carbon disulfide	secondary	5.91e-11	2.99e-12
CRT Incineration	o-xylene	secondary	5.21e-11	2.63e-12
CRT landfilling	Chromium (III)	primary	4.83e-11	2.44e-12
LPG Production	Benzyl chloride	secondary	4.52e-11	2.29e-12
LPG Production	Cobalt	secondary	4.34e-11	2.19e-12
US electric grid	Methyl tert-butyl ether	model/secondary	3.74e-11	1.89e-12
LPG Production	Aluminum (elemental)	secondary	3.34e-11	1.69e-12
LPG Production	Chloroform	secondary	3.19e-11	1.61e-12
LPG Production	Propionaldehyde	secondary	2.95e-11	1.49e-12
US electric grid	Phenol	model/secondary	2.85e-11	1.44e-12
US electric grid	Vinyl acetate	model/secondary	2.66e-11	1.34e-12
US electric grid	Phenanthrene	model/secondary	2.51e-11	1.27e-12
US electric grid	Ethylene dibromide	model/secondary	2.47e-11	1.25e-12
US electric grid	Xylene (mixed isomers)	model/secondary	2.21e-11	1.12e-12
US electric grid	Chromium (III)	model/secondary	1.97e-11	9.96e-13
LPG Production	Acrolein	secondary	1.67e-11	8.46e-13
LPG Production	Methyl chloride	secondary	1.39e-11	7.02e-13
US electric grid	2,3,7,8-TCDF	model/secondary	1.17e-11	5.90e-13
US electric grid	1,1,1-Trichloroethane	model/secondary	1.08e-11	5.48e-13
LPG Production	Di(2-ethylhexyl)phthalate	secondary	1.00e-11	5.06e-13
LPG Production	Acetaldehyde	secondary	7.97e-12	4.03e-13
US electric grid	Acenaphthene	model/secondary	7.61e-12	3.85e-13
LPG Production	Mercury	secondary	7.50e-12	3.79e-13
US electric grid	Ethyl Chloride	model/secondary	7.19e-12	3.63e-13
US electric grid	Cumene	model/secondary	6.08e-12	3.07e-13
LPG Production	Cadmium cmpds	secondary	6.00e-12	3.03e-13
LPG Production	Dimethyl sulfate	secondary	5.54e-12	2.80e-13
LPG Production	Toluene	secondary	5.38e-12	2.72e-13
US electric grid	Biphenyl	model/secondary	3.63e-12	1.83e-13
LPG Production	1,4-Dichlorobenzene	secondary	3.13e-12	1.58e-13
LPG Production	Isophorone	secondary	2.68e-12	1.35e-13
LPG Production	Aromatic hydrocarbons	secondary	2.42e-12	1.22e-13
US electric grid	Acenaphthylene	model/secondary	2.26e-12	1.14e-13
LPG Production	Methyl ethyl ketone	secondary	2.14e-12	1.08e-13
US electric grid	Benzo[a]pyrene	model/secondary	2.14e-12	1.08e-13
US electric grid	Benzo[b,j,k]fluoranthene	model/secondary	2.13e-12	1.08e-13

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
LPG Production	Tetrachloroethylene	secondary	2.12e-12	1.07e-13
US electric grid	Anthracene	model/secondary	1.97e-12	9.96e-14
LPG Production	1,2-Dichloroethane	secondary	1.88e-12	9.52e-14
LPG Production	Methyl methacrylate	secondary	1.83e-12	9.26e-14
LPG Production	Styrene	secondary	1.62e-12	8.18e-14
LPG Production	Bromoform	secondary	1.51e-12	7.64e-14
LPG Production	Dichloromethane	secondary	1.49e-12	7.54e-14
LPG Production	Chromium (III)	secondary	1.48e-12	7.49e-14
US electric grid	Benzo[a]anthracene	model/secondary	1.42e-12	7.16e-14
LPG Production	Chlorobenzene	secondary	1.21e-12	6.11e-14
LPG Production	Copper (+1 & +2)	secondary	1.13e-12	5.71e-14
US electric grid	Chrysene	model/secondary	1.03e-12	5.19e-14
US electric grid	Indeno(1,2,3-cd)pyrene	model/secondary	1.01e-12	5.09e-14
LPG Production	2,4-Dinitrotoluene	secondary	9.70e-13	4.90e-14
LPG Production	Acetophenone	secondary	8.91e-13	4.50e-14
US electric grid	Fluorene	model/secondary	8.00e-13	4.04e-14
US electric grid	Fluoranthene	model/secondary	6.39e-13	3.23e-14
LPG Production	o-xylene	secondary	5.92e-13	2.99e-14
US electric grid	Dibenzo[a,h]anthracene	model/secondary	5.48e-13	2.77e-14
US electric grid	2-Methylnaphthalene	model/secondary	5.34e-13	2.70e-14
US electric grid	Pyrene	model/secondary	5.29e-13	2.67e-14
LPG Production	Ethylbenzene	secondary	4.80e-13	2.42e-14
US electric grid	5-Methyl chrysene	model/secondary	3.95e-13	1.99e-14
US electric grid	o-xylene	model/secondary	3.80e-13	1.92e-14
US electric grid	Benzo[g,h,i]perylene	model/secondary	3.61e-13	1.82e-14
LPG Production	Methyl tert-butyl ether	secondary	2.41e-13	1.22e-14
LPG Production	Zinc (+2)	secondary	2.03e-13	1.02e-14
LPG Production	Phenanthrene	secondary	1.96e-13	9.88e-15
LPG Production	Vinyl acetate	secondary	1.71e-13	8.66e-15
LPG Production	3-Methylcholanthrene	secondary	1.66e-13	8.37e-15
LPG Production	Ethylene dibromide	secondary	1.59e-13	8.03e-15
LPG Production	2-Methylnaphthalene	secondary	1.04e-13	5.24e-15
LPG Production	Chlorine	secondary	9.19e-14	4.64e-15
LPG Production	1,1,1-Trichloroethane	secondary	6.53e-14	3.30e-15
LPG Production	Acenaphthene	secondary	6.07e-14	3.06e-15
LPG Production	Ethyl Chloride	secondary	4.63e-14	2.34e-15
LPG Production	Cumene	secondary	3.92e-14	1.98e-15
LPG Production	Benzo[a]pyrene	secondary	3.70e-14	1.87e-15
LPG Production	Biphenyl	secondary	2.34e-14	1.18e-15
LPG Production	Lead cmpds	secondary	2.01e-14	1.02e-15
LPG Production	Dibenzo[a,h]anthracene	secondary	1.92e-14	9.71e-16
LPG Production	Acenaphthylene	secondary	1.85e-14	9.33e-16
LPG Production	Anthracene	secondary	1.82e-14	9.20e-16
LPG Production	Nickel cmpds	secondary	1.70e-14	8.60e-16
LPG Production	Benzo[a]anthracene	secondary	1.65e-14	8.31e-16
LPG Production	Benzo[b,i,k]fluoranthene	secondary	1.27e-14	6.42e-16
LPG Production	Indeno(1.2.3-cd)pyrene	secondary	1.22e-14	6.18e-16
LPG Production	Chrysene	secondary	1.04e-14	5.25e-16
LPG Production	HALON-1301	secondary	8.76e-15	4.43e-16
LPG Production	Benzo[b]fluoranthene	secondary	6.58e-15	3.32e-16

 Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public	% of Total
			Toxicity (tox-kg)	
LPG Production	Mercury compounds	secondary	6.11e-15	3.09e-16
LPG Production	Fluorene	secondary	5.70e-15	2.88e-16
LPG Production	Pyrene	secondary	4.93e-15	2.49e-16
LPG Production	Benzo[g,h,i]perylene	secondary	4.79e-15	2.42e-16
LPG Production	Fluoranthene	secondary	4.65e-15	2.35e-16
LPG Production	5-Methyl chrysene	secondary	2.54e-15	1.28e-16
LPG Production	Halogenated matter (organic)	secondary	2.01e-15	1.02e-16
LPG Production	Halogenated hydrocarbons (unspecified)	secondary	5.04e-18	2.55e-19
Natural Gas Prod.	Halogenated hydrocarbons (unspecified)	secondary	-5.91e-18	-2.98e-19
CRT Incineration	Halogenated hydrocarbons (unspecified)	secondary	-8.69e-17	-4.39e-18
Fuel Oil #4 Prod.	Halogenated hydrocarbons (unspecified)	secondary	-1.43e-15	-7.22e-17
Natural Gas Prod.	Halogenated matter (organic)	secondary	-2.37e-15	-1.20e-16
Natural Gas Prod.	Mercury compounds	secondary	-7.18e-15	-3.63e-16
Natural Gas Prod.	HALON-1301	secondary	-1.03e-14	-5.20e-16
Natural Gas Prod.	Nickel cmpds	secondary	-2.00e-14	-1.01e-15
Natural Gas Prod.	Lead cmpds	secondary	-2.37e-14	-1.20e-15
CRT Incineration	Halogenated matter (organic)	secondary	-3.67e-14	-1.85e-15
CRT Incineration	HALON-1301	secondary	-1.60e-13	-8.07e-15
Natural Gas Prod.	5-Methyl chrysene	secondary	-2.56e-13	-1.29e-14
CRT Incineration	Nickel cmpds	secondary	-3.10e-13	-1.57e-14
Fuel Oil #4 Prod.	Halogenated matter (organic)	secondary	-5.72e-13	-2.89e-14
Natural Gas Prod.	Fluoranthene	secondary	-7.34e-13	-3.71e-14
Natural Gas Prod.	Fluorene	secondary	-8.22e-13	-4.15e-14
CRT Incineration	Benzo[b]fluoranthene	secondary	-1.02e-12	-5.13e-14
Fuel Oil #4 Prod.	5-Methyl chrysene	secondary	-1.06e-12	-5.35e-14
Natural Gas Prod.	Pyrene	secondary	-1.23e-12	-6.20e-14
Natural Gas Prod.	Benzo[b,j,k]fluoranthene	secondary	-1.28e-12	-6.47e-14
Natural Gas Prod.	Copper (+1 & +2)	secondary	-1.33e-12	-6.71e-14
Natural Gas Prod.	Benzo[g,h,i]perylene	secondary	-1.60e-12	-8.09e-14
Fuel Oil #4 Prod.	Mercury compounds	secondary	-1.74e-12	-8.77e-14
Fuel Oil #4 Prod.	Benzo[g,h,i]perylene	secondary	-1.74e-12	-8.79e-14
Fuel Oil #4 Prod.	Fluoranthene	secondary	-1.88e-12	-9.50e-14
Fuel Oil #4 Prod.	Pyrene	secondary	-1.91e-12	-9.67e-14
Fuel Oil #4 Prod.	Benzo[b]fluoranthene	secondary	-2.26e-12	-1.14e-13
Fuel Oil #4 Prod.	Fluorene	secondary	-2.32e-12	-1.17e-13
Natural Gas Prod.	Biphenyl	secondary	-2.35e-12	-1.19e-13
Fuel Oil #4 Prod.	HALON-1301	secondary	-2.49e-12	-1.26e-13
Natural Gas Prod.	Chrysene	secondary	-2.72e-12	-1.38e-13
CRT Incineration	Dibenzo[a,h]anthracene	secondary	-2.82e-12	-1.42e-13
Natural Gas Prod.	Aromatic hydrocarbons	secondary	-2.84e-12	-1.43e-13
Natural Gas Prod.	Benzo[b]fluoranthene	secondary	-3.19e-12	-1.61e-13
Natural Gas Prod.	Acenaphthylene	secondary	-3.51e-12	-1.77e-13
Natural Gas Prod.	Indeno(1,2,3-cd)pyrene	secondary	-3.78e-12	-1.91e-13
Natural Gas Prod.	Cumene	secondary	-3.95e-12	-1.99e-13
Fuel Oil #4 Prod.	Chrysene	secondary	-3.98e-12	-2.01e-13
Natural Gas Prod.	Anthracene	secondary	-4.08e-12	-2.06e-13
Natural Gas Prod.	Benzo[a]anthracene	secondary	-4.29e-12	-2.17e-13
Fuel Oil #4 Prod.	Indeno(1.2.3-cd)pyrene	secondary	-4.58e-12	-2.32e-13
Natural Gas Prod.	Ethyl Chloride	secondary	-4.66e-12	-2.36e-13
Fuel Oil #4 Prod.	Nickel cmpds	secondary	-4 83e-12	-2.44e-13
	· · · · · · · · · · · · · · · · · · ·			2

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Fuel Oil #4 Prod.	Benzo[b,j,k]fluoranthene	secondary	-5.29e-12	-2.67e-13
Fuel Oil #4 Prod.	Lead cmpds	secondary	-5.72e-12	-2.89e-13
Fuel Oil #4 Prod.	Benzo[a]anthracene	secondary	-6.01e-12	-3.03e-13
Fuel Oil #4 Prod.	Dibenzo[a,h]anthracene	secondary	-6.53e-12	-3.30e-13
Natural Gas Prod.	1,1,1-Trichloroethane	secondary	-6.58e-12	-3.33e-13
Natural Gas Prod.	Cadmium cmpds	secondary	-7.04e-12	-3.56e-13
Fuel Oil #4 Prod.	Anthracene	secondary	-7.15e-12	-3.61e-13
Fuel Oil #4 Prod.	Acenaphthylene	secondary	-7.42e-12	-3.75e-13
Natural Gas Prod.	Acenaphthene	secondary	-8.90e-12	-4.50e-13
Natural Gas Prod.	Dibenzo[a,h]anthracene	secondary	-8.91e-12	-4.50e-13
Fuel Oil #4 Prod.	Biphenyl	secondary	-9.73e-12	-4.92e-13
Natural Gas Prod.	Benzo[a]pyrene	secondary	-9.90e-12	-5.00e-13
Fuel Oil #4 Prod.	Benzo[a]pyrene	secondary	-1.48e-11	-7.47e-13
Natural Gas Prod.	Ethylene dibromide	secondary	-1.60e-11	-8.09e-13
Fuel Oil #4 Prod.	Cumene	secondary	-1.63e-11	-8.24e-13
Natural Gas Prod.	Vinyl acetate	secondary	-1.73e-11	-8.72e-13
CRT Incineration	2-Methylnaphthalene	secondary	-1.74e-11	-8.80e-13
Fuel Oil #4 Prod.	Ethyl Chloride	secondary	-1.93e-11	-9.74e-13
CRT Incineration	Copper (+1 & +2)	secondary	-2.06e-11	-1.04e-12
Fuel Oil #4 Prod.	Acenaphthene	secondary	-2.28e-11	-1.15e-12
Natural Gas Prod.	Methyl tert-butyl ether	secondary	-2.42e-11	-1.22e-12
Fuel Oil #4 Prod.	1,1,1-Trichloroethane	secondary	-2.72e-11	-1.37e-12
CRT Incineration	3-Methylcholanthrene	secondary	-2.78e-11	-1.41e-12
Fuel Oil #4 Prod.	Chlorine	secondary	-3.16e-11	-1.59e-12
Natural Gas Prod.	Phenanthrene	secondary	-3.53e-11	-1.79e-12
Fuel Oil #4 Prod.	2-Methylnaphthalene	secondary	-3.63e-11	-1.83e-12
CRT Incineration	Chlorine	secondary	-3.87e-11	-1.96e-12
CRT Incineration	Benzo[g,h,i]perylene	secondary	-4.06e-11	-2.05e-12
Natural Gas Prod.	Aluminum (elemental)	secondary	-4.26e-11	-2.15e-12
CRT Incineration	Aromatic hydrocarbons	secondary	-4.41e-11	-2.23e-12
CRT Incineration	Zinc (+2)	secondary	-4.48e-11	-2.26e-12
Natural Gas Prod.	Ethylbenzene	secondary	-4.85e-11	-2.45e-12
CRT Incineration	Dichlorobenzene (mixed isomers)	secondary	-4.90e-11	-2.48e-12
Natural Gas Prod.	2-Methylnaphthalene	secondary	-5.42e-11	-2.74e-12
Fuel Oil #4 Prod.	3-Methylcholanthrene	secondary	-5.80e-11	-2.93e-12
Fuel Oil #4 Prod.	Zinc (+2)	secondary	-6.42e-11	-3.24e-12
CRT Incineration	5-Methyl chrysene	secondary	-6.54e-11	-3.31e-12
Fuel Oil #4 Prod.	Ethylene dibromide	secondary	-6.62e-11	-3.34e-12
Fuel Oil #4 Prod.	Vinyl acetate	secondary	-7.14e-11	-3.61e-12
CRT Incineration	Pyrene	secondary	-7.81e-11	-3.95e-12
Fuel Oil #4 Prod.	Phenanthrene	secondary	-7.87e-11	-3.97e-12
Natural Gas Prod.	Chromium (III)	secondary	-8.08e-11	-4.08e-12
Natural Gas Prod.	3-Methylcholanthrene	secondary	-8.67e-11	-4.38e-12
Natural Gas Prod.	Acetophenone	secondary	-8.98e-11	-4.53e-12
Natural Gas Prod.	2.4-Dinitrotoluene	secondary	-9.77e-11	-4.94e-12
Fuel Oil #4 Prod.	Methyl tert-butyl ether	secondary	-1.00e-10	-5.06e-12
CRT Incineration	Fluoranthene	secondary	-1.01e-10	-5.08e-12
Natural Gas Prod.	Chlorobenzene	secondary	-1.22e-10	-6.16e-12
CRT Incineration	Fluorene	secondary	-1.29e-10	-6.51e-12
CRT Incineration	Indeno(1,2,3-cd)pyrene	secondary	-1.40e-10	-7.05e-12
		· · · · · · ·		

 Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
CRT Incineration	Chrysene	secondary	-1.50e-10	-7.59e-12
Natural Gas Prod.	Dichloromethane	secondary	-1.50e-10	-7.60e-12
Natural Gas Prod.	Bromoform	secondary	-1.52e-10	-7.70e-12
Natural Gas Prod.	Barium cmpds	secondary	-1.61e-10	-8.13e-12
Natural Gas Prod.	Styrene	secondary	-1.63e-10	-8.24e-12
CRT Incineration	Benzo[a]anthracene	secondary	-1.83e-10	-9.23e-12
Natural Gas Prod.	Methyl methacrylate	secondary	-1.85e-10	-9.33e-12
Natural Gas Prod.	1,2-Dichloroethane	secondary	-1.90e-10	-9.60e-12
Fuel Oil #4 Prod.	Ethylbenzene	secondary	-1.99e-10	-1.01e-11
Natural Gas Prod.	Tetrachloroethylene	secondary	-2.14e-10	-1.08e-11
Natural Gas Prod.	Methyl ethyl ketone	secondary	-2.16e-10	-1.09e-11
Fuel Oil #4 Prod.	o-xylene	secondary	-2.16e-10	-1.09e-11
Natural Gas Prod.	Isophorone	secondary	-2.70e-10	-1.36e-11
Fuel Oil #4 Prod.	Chromium (III)	secondary	-3.09e-10	-1.56e-11
CRT Incineration	Anthracene	secondary	-3.17e-10	-1.60e-11
Fuel Oil #4 Prod.	Copper (+1 & +2)	secondary	-3.21e-10	-1.62e-11
CRT Incineration	Benzo[b,j,k]fluoranthene	secondary	-3.27e-10) -1.65e-11
CRT Incineration	Benzo[a]pyrene	secondary	-3.57e-10	-1.80e-11
Fuel Oil #4 Prod.	Acetophenone	secondary	-3.71e-10	-1.87e-11
CRT Incineration	Acenaphthylene	secondary	-3.72e-10) -1.88e-11
Natural Gas Prod.	Silicon	secondary	-3.72e-10) -1.88e-11
Fuel Oil #4 Prod.	2,4-Dinitrotoluene	secondary	-4.04e-10	-2.04e-11
Natural Gas Prod.	Chlorine	secondary	-4.75e-10	-2.40e-11
Fuel Oil #4 Prod.	Chlorobenzene	secondary	-5.04e-10	-2.55e-11
Natural Gas Prod.	Zinc (+2)	secondary	-5.26e-10	-2.66e-11
Natural Gas Prod.	Dimethyl sulfate	secondary	-5.59e-10	-2.82e-11
CRT Incineration	Biphenyl	secondary	-6.02e-10	-3.04e-11
Fuel Oil #4 Prod.	Dichloromethane	secondary	-6.22e-10	-3.14e-11
Fuel Oil #4 Prod.	Bromoform	secondary	-6.30e-10	-3.18e-11
Fuel Oil #4 Prod.	Styrene	secondary	-6.75e-10	-3.41e-11
Fuel Oil #4 Prod.	Aromatic hydrocarbons	secondary	-6.87e-10	-3.47e-11
Fuel Oil #4 Prod.	Methyl methacrylate	secondary	-7.64e-10	-3.86e-11
Fuel Oil #4 Prod.	1,2-Dichloroethane	secondary	-7.85e-10	-3.97e-11
Natural Gas Prod.	Acetaldehyde	secondary	-8.03e-10	-4.06e-11
Fuel Oil #4 Prod.	Tetrachloroethylene	secondary	-8.83e-10) -4.46e-11
Fuel Oil #4 Prod.	Methyl ethyl ketone	secondary	-8.93e-10	-4.51e-11
Natural Gas Prod.	Mercury	secondary	-9.44e-10) -4.77e-11
CRT Incineration	Cumene	secondary	-1.01e-09	-5.09e-11
Natural Gas Prod.	Di(2-ethylhexyl)phthalate	secondary	-1.01e-09	-5.10e-11
CRT Incineration	Acenaphthene	secondary	-1.02e-09	-5.14e-11
CRT Incineration	Aluminum (elemental)	secondary	-1.08e-09	-5.47e-11
Fuel Oil #4 Prod.	1.4-Dichlorobenzene	secondary	-1.09e-09	-5.53e-11
Fuel Oil #4 Prod.	Isophorone	secondary	-1.12e-09	-5.64e-11
CRT Incineration	Ethyl Chloride	secondary	-1.19e-09	-6.02e-11
Natural Gas Prod.	Methyl chloride	secondary	-1.40e-09	-7.07e-11
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	-1.64e-09	-8.27e-11
CRT Incineration	1,1,1-Trichloroethane	secondary	-1.68e-09	-8.50e-11
Natural Gas Prod.	Acrolein	secondary	-1.69e-09	-8.53e-11
Fuel Oil #4 Prod.	Cadmium cmpds	secondary	-1.70e-09	-8.60e-11
Fuel Oil #4 Prod.	Toluene	secondary	-2.04e-09	-1.03e-10
		2		

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
CRT Incineration	Xylene (mixed isomers)	secondary	-2.13e-09	-1.08e-10
Natural Gas Prod.	o-xylene	secondary	-2.31e-09	-1.17e-10
Fuel Oil #4 Prod.	Dimethyl sulfate	secondary	-2.31e-09	-1.17e-10
Natural Gas Prod.	Cobalt	secondary	-2.65e-09	-1.34e-10
Natural Gas Prod.	Aluminum (+3)	secondary	-2.87e-09	-1.45e-10
Natural Gas Prod.	Propionaldehyde	secondary	-2.97e-09	-1.50e-10
Fuel Oil #4 Prod.	Mercury	secondary	-3.11e-09	-1.57e-10
Natural Gas Prod.	Chloroform	secondary	-3.21e-09	-1.62e-10
Fuel Oil #4 Prod.	Acetaldehyde	secondary	-3.32e-09	-1.68e-10
CRT Incineration	Phenanthrene	secondary	-4.02e-09	-2.03e-10
CRT Incineration	Ethylene dibromide	secondary	-4.09e-09	-2.07e-10
Fuel Oil #4 Prod.	Di(2-ethylhexyl)phthalate	secondary	-4.17e-09	-2.11e-10
Natural Gas Prod.	Toluene	secondary	-4.33e-09	-2.19e-10
CRT Incineration	Vinyl acetate	secondary	-4.41e-09	-2.23e-10
Natural Gas Prod.	Benzyl chloride	secondary	-4.56e-09	-2.30e-10
Natural Gas Prod.	PM-10	secondary	-4.95e-09	-2.50e-10
Fuel Oil #4 Prod.	Methyl chloride	secondary	-5.79e-09	-2.92e-10
Natural Gas Prod.	Carbon disulfide	secondary	-5.95e-09	-3.01e-10
CRT Incineration	Methyl tert-butyl ether	secondary	-6.19e-09	-3.13e-10
Fuel Oil #4 Prod.	Acrolein	secondary	-6.98e-09	-3.53e-10
Natural Gas Prod.	Phenol	secondary	-7.32e-09	-3.70e-10
Natural Gas Prod.	Cadmium	secondary	-8.00e-09	-4.04e-10
CRT Incineration	Silicon	secondary	-9.47e-09	-4.78e-10
CRT Incineration	Ethylbenzene	secondary	-1.18e-08	-5.96e-10
Fuel Oil #4 Prod.	Propionaldehyde	secondary	-1.23e-08	-6.21e-10
Fuel Oil #4 Prod.	Chloroform	secondary	-1.33e-08	-6.72e-10
Natural Gas Prod.	Cyanide (-1)	secondary	-1.45e-08	-7.35e-10
Natural Gas Prod.	Nitrate	secondary	-1.51e-08	-7.62e-10
Fuel Oil #4 Prod.	Aluminum (elemental)	secondary	-1.61e-08	-8.13e-10
CRT Incineration	Chromium (III)	secondary	-1.69e-08	-8.55e-10
Fuel Oil #4 Prod.	Cobalt	secondary	-1.85e-08	-9.36e-10
Fuel Oil #4 Prod.	Benzyl chloride	secondary	-1.88e-08	-9.52e-10
Natural Gas Prod.	Manganese	secondary	-1.98e-08	-1.00e-09
CRT Incineration	Chloroacetophenone	secondary	-2.08e-08	-1.05e-09
Natural Gas Prod.	Beryllium	secondary	-2.14e-08	-1.08e-09
Natural Gas Prod.	Lead	secondary	-2.17e-08	-1.10e-09
CRT Incineration	Acetophenone	secondary	-2.29e-08	-1.16e-09
Fuel Oil #4 Prod.	Carbon disulfide	secondary	-2.46e-08	-1.24e-09
CRT Incineration	2.4-Dinitrotoluene	secondary	-2.50e-08	-1.26e-09
Natural Gas Prod.	Copper	secondary	-2.65e-08	-1.34e-09
Natural Gas Prod.	Bromomethane	secondary	-2.77e-08	-1.40e-09
Natural Gas Prod.	2-Chloroacetophenone	secondary	-2.80e-08	-1.42e-09
Natural Gas Prod.	Barium	secondary	-2.96e-08	-1.49e-09
Natural Gas Prod.	Methyl hydrazine	secondary	-2.99e-08	-1.51e-09
CRT Incineration	Chlorobenzene	secondary	-3.11e-08	-1.57e-09
Fuel Oil #4 Prod.	Cadmium	secondary	-3.26e-08	-1.65e-09
CRT Incineration	Dichloromethane	secondary	-3.82e-08	-1.93e-09
Fuel Oil #4 Prod.	Barium cmpds	secondary	-3.89e-08	-1.97e-09
CRT Incineration	Bromoform	secondary	-3.89e-08	-1.97e-09
CRT Incineration	Styrene	secondary	-4.17e-08	-2.11e-09
	•	-		

Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
CRT Incineration	Aluminum (+3)	secondary	-4.46e-08	-2.25e-09
CRT Incineration	Methyl methacrylate	secondary	-4.72e-08	-2.38e-09
CRT Incineration	1,2-Dichloroethane	secondary	-4.73e-08	-2.39e-09
CRT Incineration	Dimethylbenzanthracene	secondary	-4.82e-08	-2.44e-09
CRT Incineration	Tetrachloroethylene	secondary	-5.33e-08	-2.69e-09
Fuel Oil #4 Prod.	Barium	secondary	-5.46e-08	-2.76e-09
CRT Incineration	Methyl ethyl ketone	secondary	-5.52e-08	-2.79e-09
Fuel Oil #4 Prod.	Cyanide (-1)	secondary	-6.02e-08	-3.04e-09
Natural Gas Prod.	Naphthalene	secondary	-6.61e-08	-3.34e-09
CRT Incineration	Toluene	secondary	-6.65e-08	-3.36e-09
Natural Gas Prod.	Hydrogen sulfide	secondary	-6.75e-08	-3.41e-09
CRT Incineration	Isophorone	secondary	-6.90e-08	-3.49e-09
Fuel Oil #4 Prod.	Beryllium	secondary	-7.66e-08	-3.87e-09
Fuel Oil #4 Prod.	Manganese	secondary	-8.00e-08	-4.04e-09
Fuel Oil #4 Prod.	Naphthalene	secondary	-8.11e-08	-4.10e-09
Natural Gas Prod.	Nickel	secondary	-8.75e-08	-4.42e-09
Fuel Oil #4 Prod.	Lead	secondary	-8.90e-08	-4.50e-09
Fuel Oil #4 Prod.	Dimethylbenzanthracene	secondary	-9.64e-08	-4.87e-09
CRT Incineration	Mercury	secondary	-1.12e-07	-5.64e-09
Fuel Oil #4 Prod.	Bromomethane	secondary	-1.15e-07	-5.79e-09
Fuel Oil #4 Prod.	2-Chloroacetophenone	secondary	-1.16e-07	-5.85e-09
CRT Incineration	Phenol	secondary	-1.17e-07	-5.92e-09
Fuel Oil #4 Prod.	Methyl hydrazine	secondary	-1.24e-07	-6.25e-09
CRT Incineration	PM-10	secondary	-1.26e-07	-6.36e-09
Fuel Oil #4 Prod.	Copper	secondary	-1.34e-07	-6.77e-09
Fuel Oil #4 Prod.	Silicon	secondary	-1.41e-07	-7.10e-09
CRT Incineration	Dimethyl sulfate	secondary	-1.43e-07	-7.21e-09
Natural Gas Prod.	Dimethylbenzanthracene	secondary	-1.44e-07	-7.28e-09
CRT Incineration	Acetaldehyde	secondary	-2.05e-07	-1.04e-08
Natural Gas Prod.	Antimony	secondary	-2.24e-07	-1.13e-08
CRT Incineration	Di(2-ethylhexyl)phthalate	secondary	-2.58e-07	-1.30e-08
Fuel Oil #4 Prod.	Nitrate	secondary	-3.15e-07	-1.59e-08
CRT Incineration	Methyl chloride	secondary	-3.58e-07	-1.81e-08
CRT Incineration	Cobalt	secondary	-3.83e-07	-1.94e-08
CRT Incineration	Copper	secondary	-4.12e-07	-2.08e-08
CRT Incineration	Acrolein	secondary	-4.31e-07	-2.18e-08
Fuel Oil #4 Prod.	Antimony	secondary	-6.39e-07	-3.23e-08
CRT Incineration	Cadmium	secondary	-6.89e-07	-3.48e-08
Fuel Oil #4 Prod.	Aluminum (+3)	secondary	-6.94e-07	-3.51e-08
Natural Gas Prod.	Chromium (VI)	secondary	-7.42e-07	-3.75e-08
CRT Incineration	Propionaldehyde	secondary	-7.59e-07	-3.84e-08
CRT Incineration	Chloroform	secondary	-8.07e-07	-4.08e-08
Natural Gas Prod.	Phosphorus (yellow or white)	secondary	-8.87e-07	-4.48e-08
CRT Incineration	Pentane	secondary	-9.44e-07	-4.77e-08
Natural Gas Prod.	Molybdenum	secondary	-1.16e-06	-5.86e-08
CRT Incineration	Benzyl chloride	secondary	-1.16e-06	-5.88e-08
CRT Incineration	Hydrogen sulfide	secondary	-1.22e-06	-6.17e-08
CRT Incineration	Barium	secondary	-1.31e-06	-6.64e-08
CRT Incineration	Hexane	secondary	-1.51e-06	-7.61e-08
CRT Incineration	Carbon disulfide	secondary	-1.52e-06	-7.69e-08

Fuel Oil 44 Prod. Nickel secondary -1.46-06 +.831-0.68 Fuel Oil 44 Prod. HydrnUnorin aid secondary -1.75c-06 +.886-08 Fuel Oil 44 Prod. PM-10 secondary -1.87c-06 +.886-08 Fuel Oil 44 Prod. Pertane secondary -1.87c-06 +.934c-08 CRT Incinention Naphtabine secondary -2.25c-06 -1.14c-07 Fuel Oil 44 Prod. Pertane secondary -2.25c-06 -1.14c-07 Fuel Oil 44 Prod. CRT Incinention Pertane secondary -2.72c-06 -1.38c-07 Fuel Oil 44 Prod. Chrominim (V1) secondary -3.05c-06 -1.48c-07 Natural Gas Prod. Pertane secondary -3.05c-06 -1.48c-07 CRT Incinention Cynic (-1) secondary -3.05c-06 -1.88c-07 CRT Incinention Cynic (-1) secondary -4.05c-06 -2.15c-07 CRT Incinention Maganese secondary -4.05c-06 -2.15c-07 CRT Incinention Maganese <td< th=""><th>Process Group</th><th>Material</th><th>LCI Data Type</th><th>Chronic Public Toxicity (tox-kg)</th><th>% of Total</th></td<>	Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Natural Gas Prod. Hydrofluorie aid. secondary -1.75-66 +.832-68 Fad Ol II 4 Prod. Phenol secondary -1.875-66 -4.85-68 Fad Ol II 4 Prod. Phenol secondary -1.975-66 -4.955-66 Fad Ol II 4 Prod. Pertance secondary -2.055-66 -1.04-67 CRT Incincation Bhane secondary -2.255-66 -1.138-67 Fad Ol II 4 Prod. Cronnium (VI) secondary -2.246-66 -1.446-07 Fad Ol II 4 Prod. Cronnium (VI) secondary -3.66-66 -1.882-67 Fad Ol II 4 Prod. Cronnium (VI) secondary -3.66-66 -1.882-67 CRT Incincation Nitrati Gas Prod. Persana secondary -3.66-66 -1.882-67 CRT Incincation Cyairde (-1) secondary -3.66-66 -2.15-67 Natural Gas Prod. Hocane secondary -4.69-66 -2.37-67 CRT Incincation Maganese secondary -4.69-66 -2.37-67 CRT Incincration Magrofluorira acid	Fuel Oil #4 Prod.	Nickel	secondary	-1.64e-06	-8.31e-08
Fuel Oil 44 Prod. Phenol secondary -1.75c-06 -3.83c-08 Fuel Oil 44 Prod. Pertune secondary -1.97c-06 -3.93c-08 CRT Incincention Naphtalene secondary -2.25c-06 -1.104c-07 CRT Incincention Ehane secondary -2.25c-06 -1.104c-07 Fuel Oil 44 Prod. Incare secondary -2.38c-07 -1.404c-07 Fuel Oil 44 Prod. Pertune secondary -3.36c-06 -1.80c-07 Natural Gas Prod. Pertune secondary -3.36c-06 -1.80c-07 CRT Incincention Cynide (-1) secondary -3.36c-06 -1.88c-07 Natural Gas Prod. Jinc (clemental) secondary -4.36c-06 -2.15c-07 Natural Gas Prod. Zinc (clemental) secondary -4.36c-06 -2.35c-07 Picel Oil 44 Prod. Infuneration Baganese secondary -4.36c-06 -2.37c-07 Natural Gas Prod. Zinc (clemental) secondary -4.36c-06 -2.37c-07 Picel Oil 44 Prod. Molyd	Natural Gas Prod.	Hydrofluoric acid	secondary	-1.75e-06	-8.83e-08
Facl Oil # Prod. PM-10 secondary -1.87c-66 -9.85c-68 Fuel Oil # Prod. Pentane secondary -2.05c-66 -1.08e-67 CRT Incineration Pinane secondary -2.25c-66 -1.18e-67 Fuel Oil # Prod. Hexane secondary -2.25c-66 -1.38e-67 Fuel Oil # Prod. Cromium (VI) secondary -2.84c-66 -1.48e-67 Fuel Oil # Prod. Pontane secondary -2.84c-66 -1.48e-67 Fuel Oil # Prod. Pontane secondary -3.86c-66 -1.88e-67 Fuel Oil # Prod. Pontane secondary -3.66c-66 -1.88e-67 CRT Incineration Cyanic (-1) secondary -4.36c-66 -2.35c-67 CRT Incineration Berylian secondary -4.36c-66 -2.35c-67 CRT Incineration Berylian secondary -4.36c-66 -2.35c-67 CRT Incineration Magnase secondary -4.36c-66 -2.35c-67 CRT Incineration Magnase secondary -4.36c-66	Fuel Oil #4 Prod.	Phenol	secondary	-1.75e-06	-8.86e-08
Fuel Oil # Prod. Pentane secondary -1.97-e.6 -9.95-e.68 CRT Incineration Enhune secondary -2.25-e.05 -1.14-e07 Fuel Oil # Prod. Hexane secondary -2.25-e.05 -1.14-e07 Fuel Oil # Prod. Chromiun (VI) secondary -2.25-e.06 -1.48-e07 Fuel Oil # Prod. Chromiun (VI) secondary -3.56-e.06 -1.88-e07 Fuel Oil # Prod. Control (Control (Contro (Contro (Contro (Control (Control (Contro (Contro (Control (Contr	Fuel Oil #4 Prod.	PM-10	secondary	-1.87e-06	-9.45e-08
CRT Incineration Raphthalene secondary -2.08-06 -1.04-07 CRT Incineration Ethane secondary -2.72-06 -1.38-07 Fuel Oil #4 Prod. Chromium (VI) secondary -2.28-06 -1.48-07 Fuel Oil #4 Prod. Chromium (VI) secondary -2.94-06 -1.48-07 Fuel Oil #4 Prod. Zine (elemental) secondary -3.60-06 -1.88-07 CRT Incineration Nitrate secondary -3.60-06 -1.88-07 CRT Incineration Cyanide (-1) secondary -4.07-06 -2.08-07 Natural Gas Prod. Zine (elemental) secondary -4.07-06 -2.08-07 Natural Gas Prod. Eine (elemental) secondary -4.09-06 -2.37-07 CRT Incineration Beryllium secondary -4.09-06 -2.37-07 Fuel Oil #4 Prod. Ethane secondary -4.09-06 -2.37-07 Natural Gas Prod. Selonium secondary -7.08-06 -3.07-07 Natural Gas Prod. Efhane secondary <	Fuel Oil #4 Prod.	Pentane	secondary	-1.97e-06	-9.93e-08
CRT IncincrationEthanesecondary-2.25e-06-1.14e-07Fuel Oil #4 Prod.Chromiun (V1)secondary-2.84e-06-1.14e-07Natural Gas Prod.Pentanesecondary-2.94e-06-1.48e-07CRT IncincrationNitratesecondary-3.56e-06-1.88e-07CRT IncincrationCyaido (-1)secondary-3.72e-06-1.88e-07Natural Gas Prod.Hexanesecondary-4.07e-06-2.06e-07Natural Gas Prod.Hexanesecondary-4.26e-06-2.37e-07CRT IncincrationManganesesecondary-4.26e-06-2.37e-07Fuel Oil #4 Prod.Ethanesecondary-4.08e-06-2.37e-07CRT IncincrationManganesesecondary-4.08e-06-2.37e-07Fuel Oil #4 Prod.Ethanesecondary-6.49e-06-2.37e-07Natural Gas Prod.Seleniumsecondary-6.88e-06-3.07e-07Fuel Oil #4 Prod.Holybelemansecondary-6.88e-06-3.28e-07Natural Gas Prod.Ethanesecondary-7.08e-06-3.07e-07Fuel Oil #4 Prod.Holybelemansecondary-7.08e-06-3.58e-07Natural Gas Prod.Ethanesecondary-7.08e-06-3.58e-07CRT IncincrationMeltyl hydrazinesecondary-7.08e-06-3.58e-07CRT IncincrationMeltyl hydrazinesecondary-7.08e-06-3.58e-07CRT IncincrationMeltyl hydrazinesecondary-3.58e-07CRT Incincration </td <td>CRT Incineration</td> <td>Naphthalene</td> <td>secondary</td> <td>-2.05e-06</td> <td>-1.04e-07</td>	CRT Incineration	Naphthalene	secondary	-2.05e-06	-1.04e-07
Fuel Oil #4 Prod. Hexane secondary -2.72e-06 -1.38e-07 Fuel Oil #4 Prod. Chromium (VI) secondary -2.84e-06 -1.48e-07 Natural Gas Prod. Pentene secondary -3.56e-06 -1.88e-07 CRT Incineration Nitrate secondary -3.56e-06 -1.88e-07 CRT Incineration Cyanide (-1) secondary -3.72e-06 -1.88e-07 Natural Gas Prod. Hexane secondary -4.72e-06 -2.06e-07 VRT Incineration Beryllium secondary -4.26e-06 -2.32e-07 CRT Incineration Manganese secondary -4.89e-06 -2.32e-07 CRT Incineration Manganese secondary -6.09e-06 -3.07e-07 Natural Gas Prod. Selonium secondary -6.09e-06 -3.07e-07 Natural Gas Prod. Fluorides (F-) secondary -7.08e-06 -3.84e-07 Natural Gas Prod. Fluorides (F-) secondary -7.08e-06 -3.84e-07 Natural Gas Prod. Fluorides (F-) secondary </td <td>CRT Incineration</td> <td>Ethane</td> <td>secondary</td> <td>-2.25e-06</td> <td>-1.14e-07</td>	CRT Incineration	Ethane	secondary	-2.25e-06	-1.14e-07
Fuel Oil #4 Prod.Chromium (VI)secondary-2.84e-06-1.44e-07Natural Gas Prod.Pentanesecondary-2.94e-06-1.48e-07CRT IncinerationNitratesecondary-3.56e-06-1.88e-07CRT IncinerationCyanide (-1)secondary-3.72e-06-1.88e-07Natural Gas Prod.Hexanesecondary-4.07e-06-2.37e-07CRT IncinerationBerylliumsecondary-4.26e-06-2.37e-07Fuel Oil #4 Prod.Ethanesecondary-4.69e-06-2.37e-07Fuel Oil #4 Prod.Ethanesecondary-4.69e-06-2.37e-07Natural Gas Prod.Selriumsecondary-6.69e-06-3.07e-07Natural Gas Prod.Selriumsecondary-6.69e-06-3.07e-07Natural Gas Prod.Selriumsecondary-6.69e-06-3.64e-07Natural Gas Prod.Selriumsecondary-7.01e-06-3.58e-07Natural Gas Prod.Ethanesecondary-7.01e-06-3.58e-07Natural Gas Prod.Hydrofluoric acidsecondary-7.01e-06-3.58e-07CRT IncinerationMelybdenumsecondary-7.01e-06-3.58e-07Paul Oil # Prod.Hydrofluoric acidsecondary-1.61e-05-5.03e-07CRT IncinerationMelybdenumsecondary-1.61e-05-5.03e-07Natural Gas Prod.Nitrous oxidesecondary-1.62e-05-1.28e-06CRT IncinerationNickelsecondary-1.62e-05-1.28e-06CR	Fuel Oil #4 Prod.	Hexane	secondary	-2.72e-06	-1.38e-07
Natural Gas Prod.Pentanesecondary2.94e-06-1.48e-07Fuel Oli # Prod.Zinc (clemental)secondary-3.56e-061.82e-07CRT IncinerationNitratesecondary-3.72e-061.82e-07Natural Gas Prod.Hexanesecondary-4.07e-06-2.06e-07Natural Gas Prod.Zinc (clemental)secondary-4.07e-06-2.05e-07Fuel Oli # Prod.Ethanesecondary-4.56e-06-2.37e-07Fuel Oli # Prod.Ethanesecondary-4.69e-06-2.37e-07Fuel Oli # Prod.Ethanesecondary-6.09e-06-3.07e-07Fuel Oli # Prod.Molybdenumsecondary-6.09e-06-3.07e-07Fuel Oli # Prod.Molybdenumsecondary-6.82e-06-3.44e-07Natural Gas Prod.Fluorides (F-)secondary-6.82e-06-3.44e-07Natural Gas Prod.Ethanesecondary-7.08e-06-3.58e-07Fuel Oli # Prod.Hydrofluoric acidsecondary-7.08e-06-3.58e-07CRT IncinerationMethyl hydra/inesecondary-7.22e-06-3.65e-07CRT IncinerationMolybdenumsecondary-7.08e-06-3.86e-07Natural Gas Prod.Nitrous oxidesecondary-1.38e-05-6.37e-07CRT IncinerationMolybdenumsecondary-7.22e-06-3.65e-07Natural Gas Prod.Nitrous oxidesecondary-1.38e-05-5.85e-07CRT IncinerationMolybdenumsecondary-1.38e-05-1.38e-06 </td <td>Fuel Oil #4 Prod.</td> <td>Chromium (VI)</td> <td>secondary</td> <td>-2.84e-06</td> <td>-1.44e-07</td>	Fuel Oil #4 Prod.	Chromium (VI)	secondary	-2.84e-06	-1.44e-07
Fuel Oil #4 Pred.Zinc (elemental)secondary-3.56e-06-1.80e-07CRT IncinerationNitratesecondary-3.60e-06-1.82e-07Natural Gas Prod.Hexanesecondary-4.07e-06-2.06e-07Natural Gas Prod.Zinc (elemental)secondary-4.26e-06-2.37e-07CRT IncinerationBeylliumsecondary-4.69e-06-2.37e-07CRT IncinerationManganesesecondary-4.69e-06-2.37e-07CRT IncinerationManganesesecondary-6.09e-06-3.07e-07Natural Gas Prod.Seleniumsecondary-6.09e-06-3.07e-07Valural Gas Prod.Fluorides (F-)secondary-6.69e-06-3.28e-07Natural Gas Prod.Fluorides (F-)secondary-6.09e-06-3.28e-07Natural Gas Prod.Fluorides (F-)secondary-7.08e-06-3.58e-07CRT IncinerationBromomethanesecondary-7.08e-06-3.58e-07CRT IncinerationMethyl hydrazinesecondary-7.08e-06-3.58e-07CRT IncinerationMolybdenumsecondary-1.16e-05-5.88e-07CRT IncinerationMolybdenumsecondary-1.16e-05-5.88e-07CRT IncinerationMolybdenumsecondary-1.16e-05-5.88e-07CRT IncinerationMolybdenumsecondary-1.16e-05-5.88e-07CRT IncinerationMolybdenumsecondary-1.16e-05-5.88e-07CRT IncinerationMolybdenumsecondary-1.16e-05 <t< td=""><td>Natural Gas Prod.</td><td>Pentane</td><td>secondary</td><td>-2.94e-06</td><td>-1.48e-07</td></t<>	Natural Gas Prod.	Pentane	secondary	-2.94e-06	-1.48e-07
CRT IncinerationNitratesecondary-3.60e-06-1.82e-07CRT IncinerationCyanide (1)secondary-3.72e-06-1.88e-07Natural Gas Prod.Zinc (elemental)secondary-4.26e-06-2.06e-07CRT IncinerationBerylliumsecondary-4.59e-06-2.37e-07CRT IncinerationManganesesecondary-4.69e-06-2.37e-07CRT IncinerationManganesesecondary-4.69e-06-2.37e-07CRT IncinerationManganesesecondary-6.69e-06-3.27e-07Natural Gas Prod.Seloniumsecondary-6.69e-06-3.28e-07Natural Gas Prod.Fluorides (F-)secondary-6.69e-06-3.28e-07Natural Gas Prod.Ethanesecondary-7.01e-06-3.58e-07CRT IncinerationBerlyndruöria caidsecondary-7.02e-06-3.68e-07CRT IncinerationMethyl hydrazinesecondary-7.64e-06-3.88e-07CRT IncinerationMethyl hydrazinesecondary-1.64e-05-5.87e-07Natural Gas Prod.Nitrous oxidesecondary-1.38e-05-6.70e-06CRT IncinerationMethyl hydrazinesecondary-1.38e-05-6.70e-07Natural Gas Prod.Formaldehydesecondary-1.38e-05-6.70e-07Natural Gas Prod.Formaldehydesecondary-1.39e-05-7.02e-06CRT IncinerationNiteklsecondary-1.39e-05-1.70e-06CRT IncinerationAnimoniasecondary-2.34e-05<	Fuel Oil #4 Prod.	Zinc (elemental)	secondary	-3.56e-06	-1.80e-07
CRT IncinerationCyanide (-1)secondary-3.72e-06-1.88e-07Natural Gas Prod.Zinc (elemental)secondary-4.05e-06-2.05e-07CRT IncinerationBerylliumsecondary-4.59e-06-2.37e-07CRT IncinerationManganesesecondary-4.69e-06-2.37e-07CRT IncinerationManganesesecondary-4.69e-06-2.37e-07Natural Gas Prod.Seleniumsecondary-6.09e-06-3.07e-07Natural Gas Prod.Fluorides (F-)secondary-6.09e-06-3.28e-07Natural Gas Prod.Ethanesecondary-7.01e-06-3.54e-07Natural Gas Prod.Ethanesecondary-7.01e-06-3.54e-07CRT IncinerationBronomethanesecondary-7.08e-06-3.68e-07CRT IncinerationMolybdenumsecondary-7.08e-06-3.68e-07CRT IncinerationMolybdenumsecondary-7.54e-06-3.68e-07CRT IncinerationMolybdenumsecondary-1.18e-05-6.97e-07Natural Gas Prod.Nitrous oxidesecondary-1.18e-05-6.97e-07CRT IncinerationNickelsecondary-1.18e-05-6.97e-07CRT IncinerationNickelsecondary-1.18e-05-6.97e-07CRT IncinerationNitrous oxidesecondary-1.18e-05-1.90e-06CRT IncinerationAimonysecondary-1.18e-05-1.90e-06CRT IncinerationAimonysecondary-1.18e-05-1.90e-05	CRT Incineration	Nitrate	secondary	-3.60e-06	-1.82e-07
Natural Gas Prod.Hexanesecondary4.07e-062.06e-07Natural Gas Prod.Zinc (elemental)secondary4.52e-062.32e-07Fuel Oil #4 Prod.Ethanesecondary4.69e-062.37e-07CRT IncinerationMarganesesecondary4.69e-062.37e-07Turul Gas Prod.Seleniumsecondary-6.09e-063.07e-07Fuel Oil #4 Prod.Molybdenumsecondary-6.09e-06-3.07e-07Fuel Oil #4 Prod.Molybdenumsecondary-6.82e-06-3.28e-07Natural Gas Prod.Ethanesecondary-7.01e-06-3.54e-07Natural Gas Prod.Ethanesecondary-7.08e-06-3.58e-07CRT IncinerationBromomethanesecondary-7.08e-06-3.58e-07CRT IncinerationMelyh hydrazinesecondary-7.22e-06-3.58e-07CRT IncinerationMolybdenumsecondary-7.22e-06-5.03e-07Natural Gas Prod.Nitrous oxidesecondary-1.36e-05-5.03e-07CRT IncinerationNickelsecondary-1.36e-05-7.02e-07Natural Gas Prod.Formaldehydesecondary-1.36e-05-7.02e-07Natural Gas Prod.Formaldehydesecondary-2.16e-05-1.28e-06CRT IncinerationNickelsecondary-2.5e-05-1.28e-06CRT IncinerationZinc (elemental)secondary-2.5e-05-1.28e-06Fuel Oil #4 Prod.Hydrogen stilfdesecondary-2.5e-05-1.28e-06 <t< td=""><td>CRT Incineration</td><td>Cyanide (-1)</td><td>secondary</td><td>-3.72e-06</td><td>-1.88e-07</td></t<>	CRT Incineration	Cyanide (-1)	secondary	-3.72e-06	-1.88e-07
Natural Gas Prod.Zinc (elemental)secondary4.26e-062.215e-07CRT IncinerationBerylliumsecondary4.69e-062.32e-07CRT IncinerationManganesesecondary4.69e-062.37e-07Natural Gas Prod.Seleniumsecondary-6.09e-06-3.07e-07Fuel Oil #4 Prod.Molybdenumsecondary-6.49e-06-3.28e-07Vatural Gas Prod.Fluorides (F-)secondary-6.49e-06-3.28e-07Natural Gas Prod.Ethanesecondary-7.01e-06-3.54e-07CRT IncinerationBromomethanesecondary-7.02e-06-3.54e-07CRT IncinerationMelyl hydrofluoric acidsecondary-7.02e-06-3.56e-07CRT IncinerationMelyl hydrofluoric acidsecondary-7.64e-06-3.56e-07CRT IncinerationMelyl hydrazinesecondary-1.16e-05-5.85e-07CRT IncinerationMitrous oxidesecondary-1.38e-05-6.97e-07Natural Gas Prod.Formaldehydesecondary-1.38e-05-6.97e-07Natural Gas Prod.Formaldehydesecondary-1.36e-05-1.09e-06CRT IncinerationNickelsecondary-1.38e-05-6.97e-07Natural Gas Prod.Formaldehydesecondary-1.36e-05-1.09e-06CRT IncinerationAntimoniasecondary-1.36e-05-1.09e-06CRT IncinerationAntimoniasecondary-2.56e-05-1.28e-05Fuel Oil # Prod.Hydrochloric acidsecondary <td>Natural Gas Prod.</td> <td>Hexane</td> <td>secondary</td> <td>-4.07e-06</td> <td>-2.06e-07</td>	Natural Gas Prod.	Hexane	secondary	-4.07e-06	-2.06e-07
CRT IncinerationBerylliumsecondary4.59e-06-2.32e-07Fuel Oil #4 Prod.Ethanesecondary-4.69e-06-2.37e-07CRT IncinerationManganesesecondary-6.09e-06-3.07e-07Natural Gas Prod.Scleniumsecondary-6.09e-06-3.07e-07Natural Gas Prod.Fluorides (F-)secondary-6.09e-06-3.28e-07Natural Gas Prod.Ethanesecondary-7.01e-06-3.54e-07CRT IncinerationBromomethanesecondary-7.08e-06-3.58e-07CRT IncinerationMelyhofuoria eaidsecondary-7.08e-06-3.58e-07CRT IncinerationMolybdenumsecondary-7.08e-06-5.03e-07CRT IncinerationMolybdenumsecondary-7.08e-06-5.03e-07Natural Gas Prod.Nitrous oxidesecondary-1.16e-05-5.87e-07CRT IncinerationNickelsecondary-1.38e-05-6.97e-07CRT IncinerationNickelsecondary-1.38e-05-6.97e-07CRT IncinerationAntimoniasecondary-1.38e-05-1.09e-06CRT IncinerationAntimonysecondary-2.25e-05-1.29e-06Fuel Oil #4 Prod.Fluorides (F-)secondary-2.16e-05-1.36e-06CRT IncinerationAntimonysecondary-2.25e-05-1.09e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.25e-05-1.99e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.56e-05-1.36e-06<	Natural Gas Prod.	Zinc (elemental)	secondary	-4.26e-06	-2.15e-07
Fuel Oil #4 Prod.Ethanesecondary4.69e-06-2.37e-07CRT IncinerationManganesesecondary-4.69e-06-2.47e-07Natural Gas Prod.Seleniumsecondary-6.09e-06-3.07e-07Natural Gas Prod.Fluorides (F-)secondary-6.49e-06-3.28e-07Natural Gas Prod.Ethanesecondary-7.01e-06-3.54e-07Natural Gas Prod.Ethanesecondary-7.08e-06-3.28e-07CRT IncinerationBromomethanesecondary-7.08e-06-3.58e-07CRT IncinerationMethyl hydrazinesecondary-7.08e-06-3.58e-07CRT IncinerationMethyl hydrazinesecondary-7.04e-06-3.86e-07CRT IncinerationMolybdenumsecondary-9.96e-06-5.03e-07Natural Gas Prod.Nitrous oxidesecondary-1.16e-05-5.85e-07CRT IncinerationNickelsecondary-1.38e-05-6.97e-07Yaural Gas Prod.Formaldehydesecondary-1.38e-05-6.97e-07Fuel Oil #4 Prod.Ammoniasecondary-2.16e-05-1.99e-06CRT IncinerationAntimonysecondary-2.16e-05-1.99e-06CRT IncinerationZinc (elemental)secondary-2.16e-05-1.28e-06CRT IncinerationZinc (elemental)secondary-2.16e-05-1.28e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.16e-05-1.38e-07CRT IncinerationZinc (elemental)secondary-2.16e-05 <td>CRT Incineration</td> <td>Beryllium</td> <td>secondary</td> <td>-4.59e-06</td> <td>-2.32e-07</td>	CRT Incineration	Beryllium	secondary	-4.59e-06	-2.32e-07
CRT IncinerationManganesesecondary4.89e-06-2.47e-07Natural Gas Prod.Scleniumsecondary-6.09e-06-3.07e-07Fuel Oil #4 Prod.Molybdenumsecondary-6.49e-06-3.28e-07Natural Gas Prod.Florides (F-)secondary-6.82e-06-3.54e-07Natural Gas Prod.Ethanesecondary-7.01e-06-3.54e-07CRT IncinerationBromomethanesecondary-7.08e-06-3.58e-07CRT IncinerationMethyl hydrazinesecondary-7.64e-06-3.86e-07CRT IncinerationMolybdenumsecondary-7.64e-06-3.86e-07Natural Gas Prod.Nitrous oxidesecondary-1.16e-05-5.85e-07CRT IncinerationNickelsecondary-1.38e-05-6.97e-07Natural Gas Prod.Formaldehydesecondary-1.38e-05-7.02e-07Natural Gas Prod.Formaldehydesecondary-1.39e-05-7.02e-07Fuel Oil #4 Prod.Ammoniasecondary-2.16e-05-1.28e-06CRT IncinerationZine (etemental)secondary-2.56e-05-1.28e-06CRT IncinerationZine (etemental)secondary-2.66e-05-1.34e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.66e-05-1.34e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.66e-05-1.36e-06Natural Gas Prod.Hydrochloric acidsecondary-3.98e-05-2.78e-06CRT IncinerationPhosphorus (yellow or white)<	Fuel Oil #4 Prod.	Ethane	secondary	-4.69e-06	-2.37e-07
Natural Gas Prod.Scleniumsecondary-6.09c-06-3.07c-07Fuel Oli #4 Prod.Molybdenumsecondary-6.49c-06-3.28c-07Natural Gas Prod.Ethanesecondary-7.01c-06-3.34e-07CRT IncinerationBromomethanesecondary-7.02c-06-3.58c-07CRT IncinerationMothybdenumsecondary-7.28c-06-3.58c-07CRT IncinerationMethyl hydrazinesecondary-7.64c-06-3.58c-07CRT IncinerationMolybdenumsecondary-9.96c-06-5.03c-07Natural Gas Prod.Nitrous oxidesecondary-1.16c-05-5.85c-07Natural Gas Prod.Nitrous oxidesecondary-1.39e-05-6.07c-07Natural Gas Prod.Formaldehydesecondary-1.39e-05-7.02c-07Valural Gas Prod.Formaldehydesecondary-1.38e-05-6.07c-07CRT IncinerationNickelsecondary-1.38e-05-1.09c-06CRT IncinerationAntimonysecondary-2.56c-05-1.28e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.58c-05-1.28e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.58c-05-1.28e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.58c-05-1.28e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-2.58c-05-1.28e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-3.98e-05-2.01e-06CRT IncinerationPhosphorus (yellow or white)	CRT Incineration	Manganese	secondary	-4.89e-06	-2.47e-07
Fuel Oil #4 Prod.Molybdenumsecondary-6.49c-06-3.28c-07Natural Gas Prod.Fluorides (F-)secondary-6.82c-06-3.44c-07Natural Gas Prod.Ethanesecondary-7.01c-06-3.54c-07CRT IncinerationBromomethanesecondary-7.28c-06-3.58c-07Fuel Oil #4 Prod.Hydrofluoric acidsecondary-7.24c-06-3.58c-07CRT IncinerationMethyl hydrazinesecondary-7.64c-06-5.85c-07CRT IncinerationMolybdenumsecondary-9.96e-06-5.03c-07Natural Gas Prod.Nitrous oxidesecondary-1.16c-05-5.55c-07CRT IncinerationNickelsecondary-1.39c-05-7.02c-07Natural Gas Prod.Formaldehydesecondary-1.39c-05-7.02c-07CRT IncinerationNickelsecondary-1.39c-05-7.02c-07CRT IncinerationAntimonysecondary-2.16c-05-1.99c-06CRT IncinerationZine (clemental)secondary-2.16c-05-1.28c-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.56c-05-1.28c-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-3.28c-05-1.36c-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-3.28c-05-1.36c-06Natural Gas Prod.Hydrochloric acidsecondary-3.28c-05-1.36c-06CRT IncinerationPhosphorus (yellow or white)secondary-3.98c-05-2.18c-06CRT IncinerationPhosphoru	Natural Gas Prod.	Selenium	secondary	-6.09e-06	-3.07e-07
Natural Gas Prod.Fluorides (F-)secondary-6.82e-06-3.44e-07Natural Gas Prod.Ethanesecondary-7.01e-06-3.54e-07CRT IncinerationBromomethanesecondary-7.08e-06-3.58e-07Fuel Oil #4 Prod.Hydrofluoric acidsecondary-7.64e-06-3.86e-07CRT IncinerationMethyl hydrazinesecondary-7.64e-06-3.86e-07CRT IncinerationMolybdenumsecondary-9.96e-06-5.03e-07Natural Gas Prod.Nitrous oxidesecondary-1.16e-05-5.85e-07CRT IncinerationNickelsecondary-1.38e-05-7.02e-07Fuel Oil #4 Prod.Formaldehydesecondary-1.38e-05-7.02e-07Fuel Oil #4 Prod.Ammoniasecondary-2.16e-05-1.09e-06CRT IncinerationZnic (elemental)secondary-2.25e-05-1.29e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.56e-05-1.36e-06Fuel Oil #4 Prod.Fluorides (F-)secondary-2.66e-05-1.36e-06Fuel Oil #4 Prod.Fluorides (F-)secondary-3.20e-05-1.36e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-3.98e-05-2.01e-06Natural Gas Prod.Hydrochloric acidsecondary-3.98e-05-2.16e-06Natural Gas Prod.Hydrochloric acidsecondary-3.38e-05-2.01e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.98e-05-2.01e-06GRT Incineration <td>Fuel Oil #4 Prod.</td> <td>Molybdenum</td> <td>secondary</td> <td>-6.49e-06</td> <td>-3.28e-07</td>	Fuel Oil #4 Prod.	Molybdenum	secondary	-6.49e-06	-3.28e-07
Natural Gas Prod.Ethanesecondary-7.01e-06-3.54e-07CRT IncinerationBromomethanesecondary-7.08e-06-3.58e-07Fuel Oil #4 Prod.Hydrofluoric acidsecondary-7.22e-06-3.65e-07CRT IncinerationMethyl hydrazinesecondary-9.96e-06-5.03e-07CRT IncinerationMolybdenumsecondary-9.96e-06-5.03e-07Natural Gas Prod.Nitrous oxidesecondary-1.16e-05-5.85e-07CRT IncinerationNickelsecondary-1.38e-05-6.07e-07Fuel Oil #4 Prod.Formaldehydesecondary-1.38e-05-7.02e-07Fuel Oil #4 Prod.Formaldehydesecondary-1.74e-05-8.79e-07CRT IncinerationAntimonysecondary-2.16e-05-1.09e-06CRT IncinerationZinc (elemental)secondary-2.56e-05-1.29e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.56e-05-1.28e-06Fuel Oil #4 Prod.Fluorides (F-)secondary-2.66e-05-1.36e-06Fuel Oil #4 Prod.Seleniumsecondary-3.20e-05-1.62e-06Natural Gas Prod.Hydrochloric acidsecondary-3.98e-05-2.01e-06CRT IncinerationPhosphorus (yellow or white)secondary-3.98e-05-2.01e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.32e-04-6.68e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-1.32e-04-6.68e-06CRT In	Natural Gas Prod.	Fluorides (F-)	secondary	-6.82e-06	-3.44e-07
CRT IncinerationBromomethanesecondary7.08e-06-3.58e-07Fuel Oil #4 Prod.Hydrofluoric acidsecondary-7.22e-06-3.65e-07CRT IncinerationMethyl hydrazinesecondary-7.64e-06-5.86e-07CRT IncinerationMolybdenumsecondary-9.96e-06-5.03e-07Natural Gas Prod.Nitrous oxidesecondary-1.16e-05-5.85e-07CRT IncinerationNickelsecondary-1.16e-05-5.85e-07CRT IncinerationNickelsecondary-1.38e-05-6.97e-07Natural Gas Prod.Formaldehydesecondary-1.74e-05-7.02e-07Fuel Oil #4 Prod.Ammoniasecondary-2.16e-05-1.09e-06CRT IncinerationAntimonysecondary-2.54e-05-1.28e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.54e-05-1.28e-06Fuel Oil #4 Prod.Fluorides (F-)secondary-2.66e-05-1.34e-06Fuel Oil #4 Prod.Seleniumsecondary-3.20e-05-1.62e-06Natural Gas Prod.Hydrochloric acidsecondary-3.29e-05-1.62e-06Natural Gas Prod.Vanadiumsecondary-3.29e-05-1.28e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.29e-05-1.28e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.29e-05-1.28e-06CRT IncinerationPhosphorus (yellow or white)secondary-3.29e-05-2.78e-06CRT Incineration <td>Natural Gas Prod.</td> <td>Ethane</td> <td>secondary</td> <td>-7.01e-06</td> <td>-3.54e-07</td>	Natural Gas Prod.	Ethane	secondary	-7.01e-06	-3.54e-07
Fuel Oil #4 Prod.Hydrofluoric acidsecondary-7.22e-06-3.65e-07CRT IncinerationMethyl hydrazinesecondary-7.64e-06-3.86e-07CRT IncinerationMolybdenumsecondary-9.96e-06-5.03e-07Natural Gas Prod.Nitrous oxidesecondary-1.16e-05-5.85e-07CRT IncinerationNickelsecondary-1.38e-05-6.97e-07Natural Gas Prod.Formaldehydesecondary-1.39e-05-7.02e-07Fuel Oil #4 Prod.Ammoniasecondary-1.74e-05-8.79e-07CRT IncinerationAntimonysecondary-2.16e-05-1.09e-06CRT IncinerationZinc (elemental)secondary-2.54e-05-1.28e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.56e-05-1.28e-06Fuel Oil #4 Prod.Fluorides (F-)secondary-2.66e-05-1.34e-06Fuel Oil #4 Prod.Seleniumsecondary-3.20e-05-1.36e-06Natural Gas Prod.Vanadiumsecondary-3.20e-05-1.62e-06Natural Gas Prod.Vanadiumsecondary-3.20e-05-1.76e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.20e-05-2.78e-06CRT IncinerationAmmoniasecondary-7.32e-04-6.24e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-1.32e-04-6.62e-06CRT IncinerationAmmoniasecondary-1.32e-04-6.62e-06Fuel Oil #4 Prod.Hydrochloric acid<	CRT Incineration	Bromomethane	secondary	-7.08e-06	-3.58e-07
CRT IncinerationMethyl hydrazinesecondary-7.64e-06-3.86e-07CRT IncinerationMolybdenumsecondary-9.96e-06-5.03e-07Natural Gas Prod.Nitrous oxidesecondary-1.16e-05-5.85e-07CRT IncinerationNickelsecondary-1.38e-05-6.97e-07Natural Gas Prod.Formaldehydesecondary-1.39e-05-7.02e-07Fuel Oil #4 Prod.Ammoniasecondary-2.16e-05-1.09e-06CRT IncinerationZinc (elemental)secondary-2.54e-05-1.28e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.55e-05-1.28e-06Fuel Oil #4 Prod.Fluorides (F-)secondary-2.66e-05-1.34e-06Fuel Oil #4 Prod.Seleniumsecondary-3.20e-05-1.62e-06Natural Gas Prod.Hydrochloric acidsecondary-3.20e-05-1.62e-06Natural Gas Prod.Vanadiumsecondary-3.20e-05-1.62e-06Natural Gas Prod.Vanadiumsecondary-3.20e-05-1.62e-06CRT IncinerationPhosphorus (yellow or white)secondary-3.20e-05-2.78e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.20e-05-2.78e-06CRT IncinerationAmmoniasecondary-1.32e-04-6.24e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.32e-04-6.24e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.32e-04-6.24e-06Fuel Oil #4 Prod.Nitrou	Fuel Oil #4 Prod.	Hydrofluoric acid	secondary	-7.22e-06	-3.65e-07
CRT IncinerationMolybdenumsecondary-9.96e-06-5.03e-07Natural Gas Prod.Nitrous oxidesecondary-1.16e-05-5.85e-07CRT IncinerationNickelsecondary-1.38e-05-6.97e-07Natural Gas Prod.Formaldehydesecondary-1.39e-05-7.02e-07Fuel Oil #4 Prod.Ammoniasecondary-1.74e-05-8.79e-07CRT IncinerationAntimonysecondary-2.16e-05-1.09e-06CRT IncinerationZinc (elemental)secondary-2.54e-05-1.28e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.56e-05-1.29e-06Fuel Oil #4 Prod.Fluorides (F-)secondary-2.66e-05-1.34e-06Fuel Oil #4 Prod.Seleniumsecondary-2.20e-05-1.36e-06Natural Gas Prod.Hydrochloric acidsecondary-3.34e-05-1.76e-06Fuel Oil #4 Prod.Seleniumsecondary-3.34e-05-1.76e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.98e-05-2.01e-06CRT IncinerationPhosphorus (yellow or white)secondary-1.32e-04-6.24e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.32e-04-6.24e-06CRT IncinerationAmmoniasecondary-1.56e-04-7.97e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.58e-04-7.97e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.58e-04-7.97e-06Fuel Oil #4 Prod.Hydro	CRT Incineration	Methyl hydrazine	secondary	-7.64e-06	-3.86e-07
Natural Gas Prod.Nitrous oxidesecondary-1.16e-05-5.85e-07CRT IncinerationNickelsecondary-1.38e-05-6.97e-07Natural Gas Prod.Formaldehydesecondary-1.39e-05-7.02e-07Fuel Oil #4 Prod.Ammoniasecondary-1.74e-05-8.79e-07CRT IncinerationAntimonysecondary-2.16e-05-1.09e-06CRT IncinerationZinc (elemental)secondary-2.54e-05-1.28e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.56e-05-1.34e-06Fuel Oil #4 Prod.Fluorides (F-)secondary-2.66e-05-1.34e-06Fuel Oil #4 Prod.Seleniumsecondary-2.69e-05-1.36e-06Natural Gas Prod.Hydrochloric acidsecondary-3.20e-05-1.62e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.98e-05-2.01e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.98e-05-2.01e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-1.32e-04-6.68e-06CRT IncinerationAmmoniasecondary-1.32e-04-6.68e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.32e-04-6.68e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.68e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.32e-04-6.68e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.68e-06Fu	CRT Incineration	Molybdenum	secondary	-9.96e-06	-5.03e-07
CRT IncinerationNickelsecondary-1.38e-05-6.97e-07Natural Gas Prod.Formaldehydesecondary-1.39e-05-7.02e-07Fuel Oil #4 Prod.Ammoniasecondary-1.74e-05-8.79e-07CRT IncinerationAntimonysecondary-2.16e-05-1.09e-06CRT IncinerationZinc (elemental)secondary-2.54e-05-1.28e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.55e-05-1.29e-06Fuel Oil #4 Prod.Fluorides (F-)secondary-2.66e-05-1.34e-06Natural Gas Prod.Hydrochloric acidsecondary-3.20e-05-1.62e-06Natural Gas Prod.Vanadiumsecondary-3.98e-05-1.62e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.98e-05-2.01e-06CRT IncinerationPhosphorus (yellow or white)secondary-3.98e-05-2.01e-06CRT IncinerationPhosphorus (yellow or white)secondary-3.98e-05-2.01e-06CRT IncinerationPhosphorus (yellow or white)secondary-1.32e-04-6.24e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.32e-04-6.24e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.32e-04-6.24e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.24e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.24e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.24e	Natural Gas Prod.	Nitrous oxide	secondary	-1.16e-05	-5.85e-07
Natural Gas Prod.Formaldehydesecondary-1.39e-05-7.02e-07Fuel Oil #4 Prod.Ammoniasecondary-1.74e-05-8.79e-07CRT IncinerationAntimonysecondary-2.16e-05-1.09e-06CRT IncinerationZinc (elemental)secondary-2.54e-05-1.28e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.55e-05-1.29e-06Fuel Oil #4 Prod.Fluorides (F-)secondary-2.66e-05-1.34e-06Fuel Oil #4 Prod.Seleniumsecondary-2.69e-05-1.36e-06Natural Gas Prod.Hydrochloric acidsecondary-3.20e-05-1.62e-06Natural Gas Prod.Vanadiumsecondary-3.98e-05-1.76e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.98e-05-2.78e-06CRT IncinerationPhosphorus (yellow or white)secondary-3.98e-05-2.78e-06CRT IncinerationAmmoniasecondary-7.13e-05-3.60e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-7.13e-05-3.60e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.24e-04-6.24e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.8e-06CRT IncinerationChromium (VI)secondary-1.56e-04-7.86e-06Natural Gas Prod.Arsenicsecondary-2.13e-04-1.08e-05Natural Gas Prod.Arsenicsecondary-2.13e-04-1.08e-05Natural Gas Prod.Ars	CRT Incineration	Nickel	secondary	-1.38e-05	-6.97e-07
Fuel Oil #4 Prod.Ammoniasecondary-1.74e-05-8.79e-07CRT IncinerationAntimonysecondary-2.16e-05-1.09e-06CRT IncinerationZinc (elemental)secondary-2.54e-05-1.28e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.55e-05-1.29e-06Fuel Oil #4 Prod.Fluorides (F-)secondary-2.66e-05-1.34e-06Fuel Oil #4 Prod.Seleniumsecondary-2.69e-05-1.36e-06Natural Gas Prod.Hydrochloric acidsecondary-3.20e-05-1.62e-06Fuel Oil #4 Prod.Vanadiumsecondary-3.98e-05-2.01e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.98e-05-2.01e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-5.50e-05-2.78e-06CRT IncinerationPhosphorus (yellow or white)secondary-7.13e-05-3.60e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.24e-04-6.24e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.68e-06CRT IncinerationChromium (VI)secondary-1.56e-04-7.86e-06Natural Gas Prod.Arsenicsecondary-2.13e-04-6.68e-06CRT IncinerationChromium (VI)secondary-2.13e-04-6.68e-06Natural Gas Prod.Arsenicsecondary-2.13e-04-1.08e-05Natural Gas Prod.Arsenicsecondary-2.13e-04-1.08e-05Natural Gas Prod	Natural Gas Prod.	Formaldehyde	secondary	-1.39e-05	-7.02e-07
CRT IncinerationAntimonysecondary-2.16e-05-1.09e-06CRT IncinerationZinc (elemental)secondary-2.54e-05-1.28e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.55e-05-1.29e-06Fuel Oil #4 Prod.Fluorides (F-)secondary-2.66e-05-1.34e-06Fuel Oil #4 Prod.Seleniumsecondary-2.69e-05-1.36e-06Natural Gas Prod.Hydrochloric acidsecondary-3.20e-05-1.62e-06Natural Gas Prod.Vanadiumsecondary-3.98e-05-2.01e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.98e-05-2.01e-06CRT IncinerationPhosphorus (yellow or white)secondary-5.50e-05-2.78e-06CRT IncinerationAmmoniasecondary-1.24e-04-6.24e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.68e-06CRT IncinerationChromium (VI)secondary-1.56e-04-7.8e-06Natural Gas Prod.Arsenicsecondary-1.32e-04-6.68e-06CRT IncinerationChromium (VI)secondary-1.32e-04-1.08e-05Natural Gas Prod.Arsenicsecondary-2.13e-04-1.08e-05Natural Gas Prod.Aumoniasecondary-2.13e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.13e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.13e-04-1.08e-05Natural Gas Prod.Sulfur	Fuel Oil #4 Prod.	Ammonia	secondary	-1.74e-05	-8.79e-07
CRT IncinerationZinc (elemental)secondary-2.54e-05-1.28e-06Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.55e-05-1.29e-06Fuel Oil #4 Prod.Fluorides (F-)secondary-2.66e-05-1.34e-06Fuel Oil #4 Prod.Seleniumsecondary-2.69e-05-1.36e-06Natural Gas Prod.Hydrochloric acidsecondary-3.20e-05-1.62e-06Natural Gas Prod.Vanadiumsecondary-3.98e-05-2.01e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.98e-05-2.01e-06CRT IncinerationPhosphorus (yellow or white)secondary-7.13e-05-3.60e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.24e-04-6.24e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.32e-04-6.68e-06CRT IncinerationAmmoniasecondary-1.56e-04-7.86e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.56e-04-7.86e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.24e-04-6.24e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.56e-04-7.86e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.56e-04-7.86e-06Ratural Gas Prod.Arsenicsecondary-2.13e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.13e-04-1.08e-05Natural Gas Prod.PMsecondary-2.16e-05-1.40e-05Natural Gas	CRT Incineration	Antimony	secondary	-2.16e-05	-1.09e-06
Fuel Oil #4 Prod.Hydrogen sulfidesecondary-2.55e-05-1.29e-06Fuel Oil #4 Prod.Fluorides (F-)secondary-2.66e-05-1.34e-06Fuel Oil #4 Prod.Seleniumsecondary-2.69e-05-1.36e-06Natural Gas Prod.Hydrochloric acidsecondary-3.20e-05-1.62e-06Natural Gas Prod.Vanadiumsecondary-3.49e-05-1.76e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.98e-05-2.01e-06CRT IncinerationPhosphorus (yellow or white)secondary-7.13e-05-3.60e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-7.13e-05-3.60e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.24e-04-6.24e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.68e-06CRT IncinerationChromium (VI)secondary-1.56e-04-7.86e-06Natural Gas Prod.Arsenicsecondary-2.13e-04-1.08e-05Natural Gas Prod.Ammoniasecondary-2.13e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.13e-04-1.08e-05Natural Gas Prod.PMsecondary-2.10e-05-2.10e-05CRT IncinerationHydrofluoric acidsecondary-2.10e-05-2.10e-05Natural Gas Prod.PMsecondary-2.10e-05-2.10e-05CRT IncinerationHydrofluoric acidsecondary-4.46e-04-2.25e-05CRT IncinerationHydro	CRT Incineration	Zinc (elemental)	secondary	-2.54e-05	-1.28e-06
Fuel Oil #4 Prod.Fluorides (F-)secondary-2.66e-05-1.34e-06Fuel Oil #4 Prod.Seleniumsecondary-2.69e-05-1.36e-06Natural Gas Prod.Hydrochloric acidsecondary-3.20e-05-1.62e-06Natural Gas Prod.Vanadiumsecondary-3.49e-05-1.76e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.98e-05-2.01e-06CRT IncinerationPhosphorus (yellow or white)secondary-5.50e-05-2.78e-06CRT IncinerationAmmoniasecondary-7.13e-05-3.60e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.24e-04-6.24e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.68e-06CRT IncinerationChromium (VI)secondary-1.56e-04-7.86e-06Natural Gas Prod.Arsenicsecondary-1.58e-04-7.97e-06Natural Gas Prod.Ammoniasecondary-2.13e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.13e-04-1.08e-05Natural Gas Prod.PMsecondary-2.16e-04-2.10e-05CRT IncinerationHydrofluoric acidsecondary-2.16e-04-2.10e-05Natural Gas Prod.PMsecondary-4.16e-04-2.25e-05CRT IncinerationHydrofluoric acidsecondary-4.46e-04-2.25e-05CRT IncinerationFormaldehydesecondary-4.47e-04-2.26e-05	Fuel Oil #4 Prod.	Hydrogen sulfide	secondary	-2.55e-05	-1.29e-06
Fuel Oil #4 Prod.Seleniumsecondary-2.69e-05-1.36e-06Natural Gas Prod.Hydrochloric acidsecondary-3.20e-05-1.62e-06Natural Gas Prod.Vanadiumsecondary-3.49e-05-1.76e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.98e-05-2.01e-06CRT IncinerationPhosphorus (yellow or white)secondary-5.50e-05-2.78e-06CRT IncinerationAmmoniasecondary-7.13e-05-3.60e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.24e-04-6.24e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.68e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.56e-04-7.86e-06Fuel Oil #4 Prod.Chromium (VI)secondary-1.58e-04-7.97e-06Natural Gas Prod.Arsenicsecondary-2.13e-04-1.08e-05Natural Gas Prod.Ammoniasecondary-2.76e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.76e-04-1.08e-05Natural Gas Prod.PMsecondary-2.76e-04-1.40e-05Natural Gas Prod.PMsecondary-2.10e-05CRT IncinerationHydrofluoric acidsecondary-2.10e-05CRT IncinerationHydrofluoric acidsecondary-4.46e-04-2.25e-05-2.25e-05-2.10e-05CRT IncinerationHydrofluoric acidsecondary-4.47e-04-2.26e-05	Fuel Oil #4 Prod.	Fluorides (F-)	secondary	-2.66e-05	-1.34e-06
Natural Gas Prod.Hydrochloric acidsecondary-3.20e-05-1.62e-06Natural Gas Prod.Vanadiumsecondary-3.49e-05-1.76e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.98e-05-2.01e-06CRT IncinerationPhosphorus (yellow or white)secondary-5.50e-05-2.78e-06CRT IncinerationAmmoniasecondary-7.13e-05-3.60e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.24e-04-6.24e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.68e-06CRT IncinerationChromium (VI)secondary-1.56e-04-7.86e-06Natural Gas Prod.Arsenicsecondary-1.58e-04-7.97e-06Natural Gas Prod.Ammoniasecondary-2.13e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.16e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.16e-04-1.08e-05Natural Gas Prod.PMsecondary-2.16e-04-1.08e-05Natural Gas Prod.PMsecondary-2.16e-04-2.10e-05CRT IncinerationHydrofluoric acidsecondary-4.15e-04-2.10e-05CRT IncinerationHydrofluoric acidsecondary-4.16e-04-2.25e-05CRT IncinerationHydrofluoric acidsecondary-4.47e-04-2.26e-05	Fuel Oil #4 Prod.	Selenium	secondary	-2.69e-05	-1.36e-06
Natural Gas Prod.Vanadiumsecondary-3.49e-05-1.76e-06Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.98e-05-2.01e-06CRT IncinerationPhosphorus (yellow or white)secondary-5.50e-05-2.78e-06CRT IncinerationAmmoniasecondary-7.13e-05-3.60e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.24e-04-6.24e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.68e-06CRT IncinerationChromium (VI)secondary-1.56e-04-7.86e-06Natural Gas Prod.Arsenicsecondary-1.58e-04-7.97e-06Natural Gas Prod.Sulfur oxidessecondary-2.13e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.76e-04-1.40e-05Natural Gas Prod.PMsecondary-4.15e-04-2.10e-05CRT IncinerationHydrofluoric acidsecondary-4.15e-04-2.25e-05CRT IncinerationFormaldehydesecondary-4.47e-04-2.25e-05	Natural Gas Prod.	Hydrochloric acid	secondary	-3.20e-05	-1.62e-06
Fuel Oil #4 Prod.Phosphorus (yellow or white)secondary-3.98e-05-2.01e-06CRT IncinerationPhosphorus (yellow or white)secondary-5.50e-05-2.78e-06CRT IncinerationAmmoniasecondary-7.13e-05-3.60e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.24e-04-6.24e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.68e-06CRT IncinerationChromium (VI)secondary-1.56e-04-7.86e-06Natural Gas Prod.Arsenicsecondary-1.58e-04-7.97e-06Natural Gas Prod.Sulfur oxidessecondary-2.13e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.76e-04-1.40e-05Natural Gas Prod.PMsecondary-4.15e-04-2.210e-05CRT IncinerationHydrofluoric acidsecondary-4.46e-04-2.25e-05CRT IncinerationFormaldehydesecondary-4.47e-04-2.26e-05	Natural Gas Prod.	Vanadium	secondary	-3.49e-05	-1.76e-06
CRT IncinerationPhosphorus (yellow or white)secondary-5.50e-05-2.78e-06CRT IncinerationAmmoniasecondary-7.13e-05-3.60e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.24e-04-6.24e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.68e-06CRT IncinerationChromium (VI)secondary-1.56e-04-7.86e-06Natural Gas Prod.Arsenicsecondary-1.58e-04-7.97e-06Natural Gas Prod.Sulfur oxidessecondary-2.13e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.76e-04-1.40e-05Natural Gas Prod.PMsecondary-4.15e-04-2.10e-05CRT IncinerationHydrofluoric acidsecondary-4.46e-04-2.25e-05CRT IncinerationFormaldehydesecondary-4.47e-04-2.26e-05	Fuel Oil #4 Prod.	Phosphorus (vellow or white)	secondary	-3.98e-05	-2.01e-06
CRT IncinerationAmmoniasecondary-7.13e-05-3.60e-06Fuel Oil #4 Prod.Nitrous oxidesecondary-1.24e-04-6.24e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.68e-06CRT IncinerationChromium (VI)secondary-1.56e-04-7.86e-06Natural Gas Prod.Arsenicsecondary-1.58e-04-7.97e-06Natural Gas Prod.Ammoniasecondary-2.13e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.76e-04-1.40e-05Natural Gas Prod.PMsecondary-4.15e-04-2.10e-05CRT IncinerationHydrofluoric acidsecondary-4.46e-04-2.25e-05CRT IncinerationFormaldehydesecondary-4.47e-04-2.26e-05	CRT Incineration	Phosphorus (yellow or white)	secondary	-5.50e-05	-2.78e-06
Fuel Oil #4 Prod.Nitrous oxidesecondary-1.24e-04-6.24e-06Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.68e-06CRT IncinerationChromium (VI)secondary-1.56e-04-7.86e-06Natural Gas Prod.Arsenicsecondary-1.58e-04-7.97e-06Natural Gas Prod.Ammoniasecondary-2.13e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.76e-04-1.40e-05Natural Gas Prod.PMsecondary-4.15e-04-2.10e-05CRT IncinerationHydrofluoric acidsecondary-4.46e-04-2.25e-05CRT IncinerationFormaldehydesecondary-4.47e-04-2.26e-05	CRT Incineration	Ammonia	secondary	-7.13e-05	-3.60e-06
Fuel Oil #4 Prod.Hydrochloric acidsecondary-1.32e-04-6.68e-06CRT IncinerationChromium (VI)secondary-1.56e-04-7.86e-06Natural Gas Prod.Arsenicsecondary-1.58e-04-7.97e-06Natural Gas Prod.Ammoniasecondary-2.13e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.76e-04-1.40e-05Natural Gas Prod.PMsecondary-4.15e-04-2.10e-05CRT IncinerationHydrofluoric acidsecondary-4.46e-04-2.25e-05CRT IncinerationFormaldehydesecondary-4.47e-04-2.26e-05	Fuel Oil #4 Prod.	Nitrous oxide	secondary	-1.24e-04	-6.24e-06
CRT IncinerationChromium (VI)secondary-1.56e-04-7.86e-06Natural Gas Prod.Arsenicsecondary-1.58e-04-7.97e-06Natural Gas Prod.Ammoniasecondary-2.13e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.76e-04-1.40e-05Natural Gas Prod.PMsecondary-4.15e-04-2.10e-05CRT IncinerationHydrofluoric acidsecondary-4.46e-04-2.25e-05CRT IncinerationFormaldehydesecondary-4.47e-04-2.26e-05	Fuel Oil #4 Prod.	Hydrochloric acid	secondary	-1.32e-04	-6.68e-06
Natural Gas Prod.Arsenicsecondary-1.58e-04-7.97e-06Natural Gas Prod.Ammoniasecondary-2.13e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.76e-04-1.40e-05Natural Gas Prod.PMsecondary-4.15e-04-2.10e-05CRT IncinerationHydrofluoric acidsecondary-4.46e-04-2.25e-05CRT IncinerationFormaldehydesecondary-4.47e-04-2.26e-05	CRT Incineration	Chromium (VI)	secondary	-1.56e-04	-7.86e-06
Natural Gas Prod.Ammoniasecondary-2.13e-04-1.08e-05Natural Gas Prod.Sulfur oxidessecondary-2.76e-04-1.40e-05Natural Gas Prod.PMsecondary-4.15e-04-2.10e-05CRT IncinerationHydrofluoric acidsecondary-4.46e-04-2.25e-05CRT IncinerationFormaldehydesecondary-4.47e-04-2.26e-05	Natural Gas Prod.	Arsenic	secondary	-1.58e-04	-7.97e-06
Natural Gas Prod.Sulfur oxidessecondary-2.76e-04-1.40e-05Natural Gas Prod.PMsecondary-4.15e-04-2.10e-05CRT IncinerationHydrofluoric acidsecondary-4.46e-04-2.25e-05CRT IncinerationFormaldehydesecondary-4.47e-04-2.26e-05	Natural Gas Prod.	Ammonia	secondary	-2.13e-04	-1.08e-05
Natural Gas Prod.PMsecondary-4.15e-04-2.10e-05CRT IncinerationHydrofluoric acidsecondary-4.46e-04-2.25e-05CRT IncinerationFormaldehydesecondary-4.47e-04-2.26e-05	Natural Gas Prod.	Sulfur oxides	secondary	-2.76e-04	-1.40e-05
CRT IncinerationHydrofluoric acidsecondary-4.46e-04-2.25e-05CRT IncinerationFormaldehydesecondary-4.47e-04-2.26e-05	Natural Gas Prod.	PM	secondary	-4.15e-04	-2.10e-05
CRT IncinerationFormaldehydesecondary-4.47e-04-2.26e-05	CRT Incineration	Hydrofluoric acid	secondary	-4.46e-04	-2.25e-05
•	CRT Incineration	Formaldehyde	secondary	-4.47e-04	-2.26e-05

Table M-33. CRT LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Fuel Oil #4 Prod.	Formaldehyde	secondary	-5.52e-04	-2.79e-05
Fuel Oil #4 Prod.	Arsenic	secondary	-7.57e-04	-3.82e-05
Fuel Oil #4 Prod.	PM	secondary	-8.70e-04	-4.40e-05
CRT Incineration	Vanadium	secondary	-1.17e-03	-5.92e-05
CRT Incineration	Selenium	secondary	-1.53e-03	-7.74e-05
CRT Incineration	Fluorides (F-)	secondary	-1.57e-03	-7.93e-05
CRT Incineration	Nitrous oxide	secondary	-2.08e-03	-1.05e-04
Fuel Oil #4 Prod.	Benzene	secondary	-2.59e-03	-1.31e-04
CRT Incineration	Benzene	secondary	-3.18e-03	-1.61e-04
Fuel Oil #4 Prod.	Nitrogen oxides	secondary	-3.86e-03	-1.95e-04
Fuel Oil #4 Prod.	Vanadium	secondary	-4.33e-03	-2.19e-04
CRT Incineration	Hydrochloric acid	secondary	-4.66e-03	-2.35e-04
Fuel Oil #4 Prod.	Sulfur oxides	secondary	-5.27e-03	-2.66e-04
Fuel Oil #4 Prod.	Methane	secondary	-5.60e-03	-2.83e-04
Natural Gas Prod.	Nitrogen oxides	secondary	-9.35e-03	-4.72e-04
CRT Incineration	Carbon monoxide	secondary	-1.56e-02	-7.88e-04
Fuel Oil #4 Prod.	Carbon monoxide	secondary	-1.63e-02	-8.23e-04
CRT Incineration	Nitrogen oxides	secondary	-2.51e-02	-1.27e-03
CRT Incineration	Methane	secondary	-3.03e-02	-1.53e-03
Natural Gas Prod.	Carbon monoxide	secondary	-3.45e-02	-1.74e-03
CRT Incineration	PM	secondary	-3.67e-02	-1.85e-03
Natural Gas Prod.	Benzene	secondary	-3.80e-02	-1.92e-03
CRT Incineration	Arsenic	secondary	-3.84e-02	-1.94e-03
Natural Gas Prod.	Methane	secondary	-5.03e-02	-2.54e-03
CRT Incineration	Sulfur oxides	secondary	-5.39e-02	-2.73e-03
Total End-of-life			1.74e-01	8.78e-03
Total All Life-cycle Stages			1.98e+03	1.00e+02

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Materials Processing Life-cycle	Stage			
Steel Prod., cold-rolled, semi-finished	Sulfur dioxide	secondary	1.24e+01	1.38e+00
PMMA Sheet Prod.	Sulfur dioxide	secondary	8.61e+00	9.55e-01
Natural Gas Prod.	Methane	secondary	7.01e+00	7.77e-01
Natural Gas Prod.	Benzene	secondary	5.29e+00	5.87e-01
Aluminum Prod.	Sulfur dioxide	secondary	5.13e+00	5.70e-01
Natural Gas Prod.	Carbon monoxide	secondary	4.80e+00	5.32e-01
Polycarbonate Production	Sulfur dioxide	secondary	4.43e+00	4.91e-01
Natural Gas Prod.	Nitrogen oxides	secondary	1.30e+00	1.44e-01
Aluminum Prod.	Titanium tetrachloride	secondary	2.68e-01	2.97e-02
Steel Prod., cold-rolled, semi-finished	Carbon monoxide	secondary	1.47e-01	1.63e-02
Steel Prod., cold-rolled, semi-finished	PM	secondary	1.03e-01	1.14e-02
Aluminum Prod.	Manganese cmpds	secondary	6.03e-02	6.69e-03
Natural Gas Prod.	PM	secondary	5.78e-02	6.42e-03
Aluminum Prod.	Vanadium	secondary	4.00e-02	4.44e-03
Natural Gas Prod.	Sulfur oxides	secondary	3.85e-02	4.27e-03
Natural Gas Prod.	Ammonia	secondary	2.97e-02	3.30e-03
PET Resin Production	Carbon monoxide	secondary	2.82e-02	3.12e-03
PMMA Sheet Prod.	Carbon monoxide	secondary	2.53e-02	2.81e-03
Steel Prod., cold-rolled, semi-finished	Vanadium	secondary	2.53e-02	2.80e-03
Polycarbonate Production	Carbon monoxide	secondary	2.50e-02	2.78e-03
PMMA Sheet Prod.	Methane	secondary	2.30e-02	2.55e-03
Polycarbonate Production	Methane	secondary	2.27e-02	2.52e-03
Natural Gas Prod.	Arsenic	secondary	2.20e-02	2.44e-03
Polycarbonate Production	Nitrogen dioxide	secondary	2.17e-02	2.40e-03
PMMA Sheet Prod.	Nitrogen dioxide	secondary	2.15e-02	2.38e-03
Aluminum Prod.	Arsenic cmpds	secondary	1.77e-02	1.96e-03
Steel Prod., cold-rolled, semi-finished	Nitrogen dioxide	secondary	1.37e-02	1.52e-03
Steel Prod., cold-rolled, semi-finished	Methane	secondary	1.07e-02	1.19e-03
Steel Prod., cold-rolled, semi-finished	Fluorides (F-)	secondary	9.41e-03	1.04e-03
PET Resin Production	Sulfur oxides	secondary	7.99e-03	8.86e-04
Styrene-butadiene Copolymer Prod.	Carbon monoxide	secondary	7.56e-03	8.39e-04
Steel Prod., cold-rolled, semi-finished	Manganese cmpds	secondary	7.30e-03	8.10e-04
Polycarbonate Production	PM	secondary	7.22e-03	8.01e-04
Aluminum Prod.	Carbon monoxide	secondary	6.93e-03	7.69e-04
Aluminum Prod.	PM	secondary	6.63e-03	7.35e-04
PMMA Sheet Prod.	PM	secondary	6.44e-03	7.15e-04
Styrene-butadiene Copolymer Prod.	Nitrogen oxides	secondary	6.11e-03	6.78e-04
Styrene-butadiene Copolymer Prod.	Methane	secondary	6.00e-03	6.66e-04
Aluminum Prod.	Methane	secondary	5.99e-03	6.65e-04
Aluminum Prod.	Nitrogen dioxide	secondary	5.31e-03	5.89e-04
Styrene-butadiene Copolymer Prod.	Sulfur oxides	secondary	4.99e-03	5.54e-04
Natural Gas Prod.	Vanadium	secondary	4.86e-03	5.39e-04
Natural Gas Prod.	Hydrochloric acid	secondary	4.46e-03	4.95e-04
Aluminum Prod.	Barium cmpds	secondary	3.89e-03	4.31e-04
PET Resin Production	Nitrogen oxides	secondary	3.45e-03	3.83e-04
Aluminum Prod.	Selenium	secondary	2.34e-03	2.60e-04
Steel Prod., cold-rolled, semi-finished	Arsenic	secondary	2.22e-03	2.46e-04
PET Resin Production	Methane	secondary	2.00e-03	2.21e-04

Table M-34. LCD LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public	% of Total
			Toxicity (tox-kg)	
Steel Prod., cold-rolled, semi-finished	Phosphorus (yellow or white)	secondary	1.99e-03	2.21e-04
Natural Gas Prod.	Formaldehyde	secondary	1.94e-03	2.15e-04
PMMA Sheet Prod.	Ammonia	secondary	1.64e-03	1.82e-04
Natural Gas Prod.	Nitrous oxide	secondary	1.61e-03	1.79e-04
PET Resin Production	PM	secondary	1.40e-03	1.55e-04
Aluminum Prod.	Aluminum (+3)	secondary	1.18e-03	1.31e-04
Steel Prod., cold-rolled, semi-finished	Hydrochloric acid	secondary	1.00e-03	1.11e-04
Natural Gas Prod.	Ethane	secondary	9.76e-04	1.08e-04
Natural Gas Prod.	Fluorides (F-)	secondary	9.49e-04	1.05e-04
Styrene-butadiene Copolymer Prod.	PM	secondary	8.68e-04	9.63e-05
Steel Prod., cold-rolled, semi-finished	Benzene	secondary	8.62e-04	9.56e-05
Steel Prod., cold-rolled, semi-finished	Barium empds	secondary	8.49e-04	9.41e-05
Natural Gas Prod.	Selenium	secondary	8.48e-04	9.40e-05
Steel Prod., cold-rolled, semi-finished	Nitrous oxide	secondary	7.62e-04	8.45e-05
Steel Prod., cold-rolled, semi-finished	Arsenic cmpds	secondary	7.53e-04	8.35e-05
Aluminum Prod.	Hydrochloric acid	secondary	6.64e-04	7.37e-05
PMMA Sheet Prod.	Nitrates/nitrites	secondary	6.47e-04	7.18e-05
Steel Prod., cold-rolled, semi-finished	Ammonia	secondary	6.19e-04	6.87e-05
Natural Gas Prod.	Zinc (elemental)	secondary	5.94e-04	6.59e-05
Natural Gas Prod.	Hexane	secondary	5.67e-04	6.29e-05
Steel Prod., cold-rolled, semi-finished	Silicon	secondary	4.96e-04	5.50e-05
Aluminum Prod.	Cadmium cmpds	secondary	4.36e-04	4.84e-05
Natural Gas Prod.	Pentane	secondary	4.09e-04	4.54e-05
Steel Prod., cold-rolled, semi-finished	Titanium tetrachloride	secondary	4.08e-04	4.53e-05
PMMA Sheet Prod.	Hydrogen cyanide	secondary	3.73e-04	4.13e-05
PMMA Sheet Prod.	Hydrochloric acid	secondary	2.99e-04	3.31e-05
Polycarbonate Production	Hydrochloric acid	secondary	2.60e-04	2.88e-05
Natural Gas Prod.	Hydrofluoric acid	secondary	2.43e-04	2.70e-05
Steel Prod., cold-rolled, semi-finished	Ethane	secondary	2.41e-04	2.67e-05
Aluminum Prod.	Benzene	secondary	2.12e-04	2.35e-05
Steel Prod., cold-rolled, semi-finished	Formaldehyde	secondary	2.03e-04	2.26e-05
Aluminum Prod.	Hydrofluoric acid	secondary	1.79e-04	1.99e-05
Polycarbonate Production	Sulfuric acid	secondary	1.77e-04	1.97e-05
Aluminum Prod.	Barium sulfate	secondary	1.69e-04	1.87e-05
Steel Prod., cold-rolled, semi-finished	Titanium	secondary	1.62e-04	1.80e-05
Natural Gas Prod.	Molybdenum	secondary	1.62e-04	1.79e-05
Steel Prod., cold-rolled, semi-finished	Selenium	secondary	1.47e-04	1.63e-05
PET Resin Production	Hydrochloric acid	secondary	1.45e-04	1.61e-05
Polycarbonate Production	Mercury compounds	secondary	1.36e-04	1.51e-05
PMMA Sheet Prod.	Sulfuric acid	secondary	1.32e-04	1.46e-05
Natural Gas Prod.	Phosphorus (vellow or white)	secondary	1.24e-04	1.37e-05
Steel Prod., cold-rolled, semi-finished	Zinc (elemental)	secondary	1.06e-04	1.18e-05
Natural Gas Prod.	Chromium (VI)	secondary	1.03e-04	1.15e-05
Aluminum Prod.	Nitrous oxide	secondary	1.03e-04	1.14e-05
Steel Prod cold-rolled semi-finished	Tin $(Sn++ Sn4+)$	secondary	1.02e-04	1 13e-05
PMMA Sheet Prod	Mercury compounds	secondary	1.01e-04	1.12e-05
Aluminum Prod	Perfluoromethane	secondary	9.65e-05	1 07e-05
Steel Prod cold-rolled semi-finished	Molybdenum	secondary	9.000 05 9.21e-05	1.07e-05
Polycarbonate Production	Aromatic hydrocarbons	secondary	9 18e-05	1.02e-05
Aluminum Prod	Titanium	secondary	8 57e-05	9.51e-06
Aluminum Prod	Zinc (elemental)	secondary	8 55e 05	9.48e-06
/ mailingin 1 100.	Zine (cicinental)	scionidary	8.556-05	2.400-00

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Aluminum Prod.	Silicon	secondary	8.14e-05	9.03e-06
Aluminum Prod.	Copper (+1 & +2)	secondary	6.70e-05	7.44e-06
Aluminum Prod.	Strontium (Sr II)	secondary	6.32e-05	7.01e-06
PET Resin Production	Sulfuric acid	secondary	6.24e-05	6.93e-06
Styrene-butadiene Copolymer Prod.	Sulfuric acid	secondary	6.22e-05	6.90e-06
Steel Prod., cold-rolled, semi-finished	Barium	secondary	6.03e-05	6.69e-06
Styrene-butadiene Copolymer Prod.	Aromatic hydrocarbons	secondary	5.93e-05	6.58e-06
Polycarbonate Production	Fluorides (F-)	secondary	5.14e-05	5.70e-06
Styrene-butadiene Copolymer Prod.	Mercury compounds	secondary	4.77e-05	5.29e-06
Steel Prod., cold-rolled, semi-finished	Antimony	secondary	4.67e-05	5.18e-06
Steel Prod., cold-rolled, semi-finished	Boron	secondary	4.20e-05	4.66e-06
Steel Prod., cold-rolled, semi-finished	Strontium (Sr II)	secondary	3.88e-05	4.31e-06
PMMA Sheet Prod.	Fluorine	secondary	3.82e-05	4.24e-06
Aluminum Prod.	Nitrate	secondary	3.40e-05	3.77e-06
Natural Gas Prod.	Antimony	secondary	3.11e-05	3.45e-06
Steel Prod., cold-rolled, semi-finished	Silver compounds	secondary	3.03e-05	3.36e-06
Styrene-butadiene Copolymer Prod.	Hydrochloric acid	secondary	2.98e-05	3.31e-06
Steel Prod., cold-rolled, semi-finished	Nitrates/nitrites	secondary	2.77e-05	3.08e-06
Steel Prod., cold-rolled, semi-finished	Chromium (VI)	secondary	2.53e-05	2.81e-06
Steel Prod., cold-rolled, semi-finished	Acetylene	secondary	2.28e-05	2.53e-06
Steel Prod., cold-rolled, semi-finished	Mercury compounds	secondary	2.22e-05	2.46e-06
Natural Gas Prod.	Dimethylbenzanthracene	secondary	2.01e-05	2.23e-06
Steel Prod cold-rolled semi-finished	Hydrofluoric acid	secondary	1 86e-05	2.06e-06
PET Resin Production	Fluorine	secondary	1.81e-05	2.01e-06
Styrene-butadiene Copolymer Prod.	Fluorides (F-)	secondary	1.80e-05	2.00e-06
Steel Prod cold-rolled semi-finished	Barium sulfate	secondary	1 68e-05	1.87e-06
Steel Prod. cold-rolled semi-finished	Aluminum (+3)	secondary	1.54e-05	1.71e-06
Aluminum Prod	Barium	secondary	1.46e-05	1.62e-06
Steel Prod cold-rolled semi-finished	Hydrogen sulfide	secondary	1 33e-05	1 48e-06
Steel Prod., cold-rolled, semi-finished	Cadmium cmpds	secondary	1.32e-05	1.47e-06
Aluminum Prod	Zinc(+2)	secondary	1 30e-05	1 44e-06
Natural Gas Prod	Nickel	secondary	1.22e-05	1 35e-06
Steel Prod cold-rolled semi-finished	Aluminum (elemental)	secondary	1.22e-05	1.34e-06
Aluminum Prod	Aromatic hydrocarbons	secondary	1.05e-05	1 17e-06
Aluminum Prod	Nickel cmpds	secondary	1.01e-05	1 12e-06
Natural Gas Prod.	Hydrogen sulfide	secondary	9.40e-06	1.04e-06
Aluminum Prod	Nickel	secondary	9 40e-06	1.04e-06
Natural Gas Prod	Nanhthalene	secondary	9.21e-06	1.02e-06
Aluminum Prod	Ammonia	secondary	9.11e-06	1.01e-06
Aluminum Prod	Copper	secondary	8 87e-06	9.84e-07
Steel Prod cold-rolled semi-finished	Pentane	secondary	8 31e-06	9.22e-07
Steel Prod. cold-rolled semi-finished	Aromatic hydrocarbons	secondary	7.02e-06	7 79e-07
PMMA Sheet Prod	Hydrofluoric acid	secondary	6 90e-06	7.66e-07
Aluminum Prod	Fluoride	secondary	6 78e-06	7 52e-07
Aluminum Prod	Lead cmpds	secondary	6.49e-06	7.32e 07
Aluminum Prod	Vanadium (V3+ V5+)	secondary	6 06e-06	6.73e-07
Steel Prod. cold-rolled semi-finished	Copper	secondary	5 99e-06	6.64e-07
Polycarbonate Production	Copper $(+1 \& +2)$	secondary	5 70e 06	6.42e-07
Steel Prod cold-rolled semi-finished	Nickel	secondary	5.770-00	6 20e-07
Steer i rou., colu-roncu, senn-mislieu	TUCKU	Secondary	5.678-00	0.290-07

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Aluminum Prod.	Nitrites	secondary	5.46e-06	6.05e-07
Aluminum Prod.	Perfluoroethane	secondary	5.36e-06	5.95e-07
Polycarbonate Production	Phenol	secondary	5.22e-06	5.79e-07
Aluminum Prod.	Chromium (VI)	secondary	5.18e-06	5.75e-07
Steel Prod., cold-rolled, semi-finished	Hexane	secondary	4.34e-06	4.82e-07
PMMA Sheet Prod.	Copper (+1 & +2)	secondary	4.31e-06	4.78e-07
Steel Prod., cold-rolled, semi-finished	Copper (+1 & +2)	secondary	4.21e-06	4.67e-07
Natural Gas Prod.	Methyl hydrazine	secondary	4.17e-06	4.62e-07
Polycarbonate Production	Hydrofluoric acid	secondary	4.13e-06	4.58e-07
Natural Gas Prod.	Barium	secondary	4.12e-06	4.57e-07
Aluminum Prod.	Aluminum (elemental)	secondary	3.92e-06	4.35e-07
Natural Gas Prod.	2-Chloroacetophenone	secondary	3.90e-06	4.33e-07
Natural Gas Prod.	Bromomethane	secondary	3.86e-06	4.28e-07
PMMA Sheet Prod.	Cyanide (-1)	secondary	3.83e-06	4.25e-07
Natural Gas Prod.	Copper	secondary	3.69e-06	4.09e-07
Alereitaren Des I	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo	1	2 (8- 0)	4.09 - 07
Aluminum Prod.	VI)	secondary	3.088-00	4.086-07
Polycarbonate Production	Ammonia	secondary	3.2/e-06	3.63e-07
PMMA Sheet Prod.	Hydrogen sulfide	secondary	3.07e-06	3.40e-07
Natural Gas Prod.	Lead	secondary	3.03e-06	3.36e-07
Natural Gas Prod.	Beryllium	secondary	2.99e-06	3.31e-07
Aluminum Prod.	Hydrogen sulfide	secondary	2.80e-06	3.11e-07
Natural Gas Prod.	Manganese	secondary	2.76e-06	3.06e-07
Aluminum Prod.	Acetic acid	secondary	2.73e-06	3.03e-07
Styrene-butadiene Copolymer Prod.	Nitrate	secondary	2.53e-06	2.81e-07
Natural Gas Prod.	Nitrate	secondary	2.10e-06	2.33e-07
Polycarbonate Production	Halogenated hydrocarbons (unspecified)	secondary	2.06e-06	2.29e-07
Polycarbonate Production	Hydrogen sulfide	secondary	2.06e-06	2.29e-07
PET Resin Production	Copper (+1 & +2)	secondary	2.04e-06	2.26e-07
Styrene-butadiene Copolymer Prod.	Copper (+1 & +2)	secondary	2.03e-06	2.25e-07
Natural Gas Prod.	Cyanide (-1)	secondary	2.03e-06	2.25e-07
Steel Prod., cold-rolled, semi-finished	Bromine	secondary	1.99e-06	2.21e-07
PET Resin Production	Hydrofluoric acid	secondary	1.82e-06	2.01e-07
PMMA Sheet Prod.	Phenol	secondary	1.75e-06	1.94e-07
Steel Prod., cold-rolled, semi-finished	Acetic acid	secondary	1.74e-06	1.93e-07
Steel Prod., cold-rolled, semi-finished	Boric acid	secondary	1.62e-06	1.80e-07
PMMA Sheet Prod.	Aromatic hydrocarbons	secondary	1.53e-06	1.70e-07
Steel Prod., cold-rolled, semi-finished	Strontium	secondary	1.47e-06	1.63e-07
Aluminum Prod.	Cadmium	secondary	1.25e-06	1.39e-07
Natural Gas Prod.	Cadmium	secondary	1.11e-06	1.24e-07
Steel Prod., cold-rolled, semi-finished	Lead	secondary	1.08e-06	1.20e-07
Steel Prod., cold-rolled, semi-finished	Triethylene glycol	secondary	1.04e-06	1.15e-07
Polycarbonate Production	Chlorine	secondary	1.03e-06	1.14e-07
Natural Gas Prod.	Phenol	secondary	1.02e-06	1.13e-07
Aluminum Prod.	Triethylene glycol	secondary	1.00e-06	1.11e-07
Steel Prod., cold-rolled, semi-finished	Cadmium	secondary	9.71e-07	1.08e-07
Aluminum Prod.	Lead	secondary	8.94e-07	9.92e-08
Steel Prod., cold-rolled semi-finished	Lead cmpds	secondary	8 65e-07	9 60e-08
Steel Prod. cold-rolled semi-finished	Uranium	secondary	8 39e-07	9.31e-08
Natural Gas Prod	Carbon disulfide	secondary	8 20e 07	9.20e-08
PMMA Sheet Prod	Chlorine	secondary	7 670 07	8.510-08
i minina bileet i iou.	Chlorine	secondary	7.078-07	0.510-08

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Steel Prod., cold-rolled, semi-finished	Toluene	secondary	7.44e-07	8.25e-08
Styrene-butadiene Copolymer Prod.	Aluminum (+3)	secondary	7.23e-07	8.02e-08
Natural Gas Prod.	PM-10	secondary	6.90e-07	7.65e-08
Aluminum Prod.	Toluene	secondary	6.75e-07	7.48e-08
PMMA Sheet Prod.	Nickel cmpds	secondary	6.48e-07	7.19e-08
Natural Gas Prod.	Benzyl chloride	secondary	6.35e-07	7.04e-08
Natural Gas Prod.	Toluene	secondary	6.03e-07	6.69e-08
Steel Prod., cold-rolled, semi-finished	Ethylene	secondary	6.02e-07	6.67e-08
Steel Prod., cold-rolled, semi-finished	Cyanide (-1)	secondary	5.92e-07	6.56e-08
Steel Prod., cold-rolled, semi-finished	Methanol	secondary	5.84e-07	6.48e-08
Steel Prod., cold-rolled, semi-finished	Xylene (mixed isomers)	secondary	5.50e-07	6.10e-08
Steel Prod., cold-rolled, semi-finished	Naphthalene	secondary	5.46e-07	6.05e-08
Steel Prod., cold-rolled, semi-finished	Zinc (+2)	secondary	5.20e-07	5.77e-08
Polycarbonate Production	Aluminum (+3)	secondary	5.16e-07	5.72e-08
Polycarbonate Production	Ethanethiol	secondary	5.16e-07	5.72e-08
Polycarbonate Production	Lead	secondary	5.16e-07	5.72e-08
Polycarbonate Production	Nitrate	secondary	5.16e-07	5.72e-08
Polycarbonate Production	Nitrous oxide	secondary	5.16e-07	5.72e-08
Polycarbonate Production	Zinc(+2)	secondary	5.16e-07	5.72e-08
Aluminum Prod.	Xylene (mixed isomers)	secondary	5.13e-07	5.69e-08
Steel Prod cold-rolled semi-finished	Phenol	secondary	4 75e-07	5.27e-08
Steel Prod cold-rolled semi-finished	Hentane	secondary	4 57e-07	5 07e-08
Natural Gas Prod	Chloroform	secondary	4 48e-07	4 97e-08
Steel Prod., cold-rolled, semi-finished	Nickel cmpds	secondary	4.15e-07	4.60e-08
Natural Gas Prod	Propionaldehyde	secondary	4 14e-07	4 59e-08
Natural Gas Prod	Aluminum (+3)	secondary	4 00e-07	4 44e-08
Steel Prod cold-rolled semi-finished	Cobalt	secondary	3 97e-07	4 40e-08
PMMA Sheet Prod	Aluminum (+3)	secondary	3 83e-07	4 25e-08
PMMA Sheet Prod	Ethanethiol	secondary	3 83e-07	4 25e-08
PMMA Sheet Prod	Lead	secondary	3 83e-07	4 25e-08
PMMA Sheet Prod	Nitrous oxide	secondary	3 83e-07	4 25e-08
PMMA Sheet Prod	Zinc (+2)	secondary	3 83e-07	4 25e-08
Natural Gas Prod	Cobalt	secondary	3 69e-07	4 09e-08
PET Resin Production	Chlorine	secondary	3 63e-07	4 03e-08
Styrene-butadiene Copolymer Prod	Chlorine	secondary	3 62e-07	4 01e-08
Styrene-butadiene Copolymer Prod	Hydrofluoric acid	secondary	3.62e-07	4 01e-08
Steel Prod cold-rolled semi-finished	Bervllium	secondary	3 53e-07	3 92e-08
Natural Gas Prod	o-xylene	secondary	3 21e-07	3 56e-08
Aluminum Prod	Cobalt	secondary	3.06e-07	3 40e-08
Steel Prod cold-rolled semi-finished	Vanadium (V3+ V5+)	secondary	2 91e-07	3 23e-08
Aluminum Prod	Strontium	secondary	2.83e-07	3.14e-08
Steel Prod cold-rolled semi-finished	Benzo[a]nvrene	secondary	2.05e 07	3.04e-08
Polycarbonate Production	Mercury	secondary	2.74e 07 2.58e-07	2.86e-08
PET Resin Production	Ammonia	secondary	2.50e 07 2.47e-07	2.00e 00
Styrene-butadiene Conolymer Prod	Ammonia	secondary	2.47e 07	2.740 08 2.73e-08
Aluminum Prod	Phenol	secondary	2.+0C=07	2.75C-08
Natural Gas Prod	Acrolein	secondary	2.360-07	2.040-08 2.61e-08
Steel Prod cold-rolled semi-finished	Boron (B III)	secondary	2.350-07	2.556-08
Natural Gas Prod	1 4-Dichlorobenzene	secondary	2.500-07	2.550-08
i vatural Gas I i Gu.	1,7-Diemorouenzene	secondary	2.200-07	2.550-00

 Table M-34.
 LCD LCIA Results for the Chronic Public Health Effects Impact Category

	Material	LCI Data Tuna	Chuonia Dublia	% of Total
Process Group	Material	LCI Data Type	Toxicity (tox-kg)	% of 1 otal
Styrene-butadiene Copolymer Prod.	Phenol	secondary	2.15e-07	2.39e-08
PET Resin Production	Phenol	secondary	1.98e-07	2.20e-08
Natural Gas Prod.	Methyl chloride	secondary	1.95e-07	2.16e-08
PMMA Sheet Prod.	Mercury	secondary	1.92e-07	2.13e-08
PET Resin Production	Mercury	secondary	1.82e-07	2.01e-08
PET Resin Production	Aluminum (+3)	secondary	1.82e-07	2.01e-08
PET Resin Production	Hydrogen sulfide	secondary	1.82e-07	2.01e-08
PET Resin Production	Lead	secondary	1.82e-07	2.01e-08
PET Resin Production	Nitrate	secondary	1.82e-07	2.01e-08
PET Resin Production	Nitrous oxide	secondary	1.82e-07	2.01e-08
PET Resin Production	Zinc (+2)	secondary	1.82e-07	2.01e-08
Styrene-butadiene Copolymer Prod.	Halogenated hydrocarbons (unspecified)	secondary	1.81e-07	2.01e-08
Styrene-butadiene Copolymer Prod.	Hydrogen sulfide	secondary	1.81e-07	2.01e-08
Styrene-butadiene Copolymer Prod.	Lead	secondary	1.81e-07	2.01e-08
Styrene-butadiene Copolymer Prod.	Nitrous oxide	secondary	1.81e-07	2.01e-08
Styrene-butadiene Copolymer Prod.	Zinc (+2)	secondary	1.81e-07	2.01e-08
Steel Prod., cold-rolled, semi-finished	Manganese	secondary	1.76e-07	1.96e-08
Natural Gas Prod.	Di(2-ethylhexyl)phthalate	secondary	1.40e-07	1.56e-08
Aluminum Prod.	HALON-1301	secondary	1.34e-07	1.49e-08
Natural Gas Prod	Mercury	secondary	1 32e-07	1 46e-08
Steel Prod cold-rolled semi-finished	Rubidium ion (Rb+)	secondary	1.19e-07	1.32e-08
Natural Gas Prod	Acetaldehyde	secondary	1.12e-07	1.24e-08
Aluminum Prod	1.2-Dichlorotetrafluoroethane	secondary	9.29e-08	1.24e 00
Styrana butadiana Conolymar Drod	Mercury	secondary	9.04e.08	1.00e.08
Styrene-butatiene coporyiner riou.	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo	secondary	9.040-08	1.000-08
Steel Prod., cold-rolled, semi-finished	VI)	secondary	8.89e-08	9.86e-09
Steel Prod., cold-rolled, semi-finished	Morpholine	secondary	8.78e-08	9.74e-09
Steel Prod., cold-rolled, semi-finished	Propylene	secondary	8.67e-08	9.62e-09
Steel Prod., cold-rolled, semi-finished	Acetaldehyde	secondary	7.93e-08	8.80e-09
Natural Gas Prod	Dimethyl sulfate	secondary	7 78e-08	8 63e-09
Aluminum Prod	Mercury	secondary	7.51e-08	8 33e-09
Natural Gas Prod	Zinc (+2)	secondary	7 32e-08	8 12e-09
Aluminum Prod	Cvanide (-1)	secondary	7.05e-08	7.82e-09
Natural Gas Prod	Chlorine	secondary	6.61e-08	7 33e-09
Steel Prod cold-rolled semi-finished	Fluorine	secondary	5.60e-08	6.21e-09
Steel Prod. cold-rolled semi-finished	Lanthanum	secondary	5.00 c 00	5.84e-09
Natural Gas Prod	Silicon	secondary	5.19e-08	5.75e-09
Aluminum Prod	Chromium (III)	secondary	4.88e-08	5.42e-09
Steel Prod cold-rolled semi-finished	Acetone	secondary	3.87e-08	4 30e-09
Natural Gas Prod	Isophorono	secondary	3.870-08	4.300-09
Natural Gas Prod	Mathyl athyl katona	secondary	3.01e.08	3 3/e 00
Natural Cas Prod	Tatuashlaro athulano	secondary	2.070.08	3.346-09
Natural Gas Frou.	The minor	secondary	2.976-08	3.300-09
Steel Prod., cold-rolled, semi-linished	1 A Dishlara Akana	secondary	2.836-08	3.13e-09
Natural Gas Prod.	1,2-Dichloroethane	secondary	2.05e-08	2.936-09
Natural Gas Prod.	Methyl methacrylate	secondary	2.57e-08	2.85e-09
Steel Prod., cold-rolled, semi-finished	Ethylbenzene	secondary	2.57e-08	2.85e-09
Steel Prod., cold-rolled, semi-finished	HALON-1301	secondary	2.47e-08	2.74e-09
Natural Gas Prod.	Styrene	secondary	2.27e-08	2.52e-09
Natural Gas Prod.	Barium empds	secondary	2.24e-08	2.49e-09
Natural Gas Prod.	Bromoform	secondary	2.12e-08	2.35e-09
Natural Gas Prod.	Dichloromethane	secondary	2.10e-08	2.32e-09

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Chlorobenzene	secondary	1.70e-08	1.88e-09
Steel Prod., cold-rolled, semi-finished	Tin	secondary	1.64e-08	1.82e-09
Steel Prod., cold-rolled, semi-finished	Mercury	secondary	1.60e-08	1.77e-09
Steel Prod., cold-rolled, semi-finished	Phosphorus pentoxide	secondary	1.58e-08	1.75e-09
Natural Gas Prod.	2,4-Dinitrotoluene	secondary	1.36e-08	1.51e-09
Steel Prod., cold-rolled, semi-finished	Hydrazine	secondary	1.30e-08	1.44e-09
Steel Prod., cold-rolled, semi-finished	Scandium	secondary	1.28e-08	1.42e-09
Natural Gas Prod.	Acetophenone	secondary	1.25e-08	1.39e-09
Natural Gas Prod.	3-Methylcholanthrene	secondary	1.21e-08	1.34e-09
Natural Gas Prod.	Chromium (III)	secondary	1.13e-08	1.25e-09
Steel Prod., cold-rolled, semi-finished	Chlorine	secondary	1.12e-08	1.24e-09
Steel Prod., cold-rolled, semi-finished	Thallium	secondary	8.72e-09	9.67e-10
Natural Gas Prod.	2-Methylnaphthalene	secondary	7.56e-09	8.38e-10
Natural Gas Prod.	Ethylbenzene	secondary	6.75e-09	7.49e-10
Steel Prod., cold-rolled, semi-finished	Zirconium	secondary	6.73e-09	7.47e-10
Natural Gas Prod	Aluminum (elemental)	secondary	5 93e-09	6 58e-10
Natural Gas Prod	Phenanthrene	secondary	4 92e-09	5.46e-10
Natural Gas Prod	Methyl tert-butyl ether	secondary	3 38e-09	3.75e-10
Steel Prod cold-rolled semi-finished	Nitrites	secondary	2.76e-09	3.06e-10
Steel Prod. cold-rolled, semi-finished	Chromium (III)	secondary	2.70e-09	2.88e-10
Natural Gas Prod	Vinyl acetate	secondary	2.5)e-09	2.686-10 2.67e-10
Natural Gas Prod	Ethylene dibromide	secondary	2.400-09	2.07C-10
Steel Bred and rolled somi finished	Educine and (EDTA)	secondary	1.800.00	2.470-10
Steel Prod., cold rolled, semi-finished	Hypochlorous acid	secondary	1.090-09	1.01e-10
Steel Prod. cold rolled, semi-finished	Cabalt (Ca L Ca II Ca III)	secondary	1.720-09	1.710-10
Natural Cas Brad	Penga[a]nymana	secondary	1.340-09	1.710-10
Natural Gas Prod.	Dihawa fa hlantha anna	secondary	1.386-09	1.556-10
Natural Gas Prod.	A seventthere	secondary	1.24e-09	1.386-10
Natural Gas Prod.	Acenaphinene	secondary	1.246-09	1.37e-10
Natural Gas Prod.		secondary	9.81e-10	1.09e-10
Natural Gas Prod.	1,1,1-1 richloroethane	secondary	9.17e-10	1.02e-10
Steel Prod., cold-rolled, semi-finished	Dichloromethane	secondary	7.33e-10	8.13e-11
Natural Gas Prod.	Ethyl Chloride	secondary	6.50e-10	7.21e-11
Natural Gas Prod.	Benzo[a]anthracene	secondary	5.98e-10	6.63e-11
Natural Gas Prod.	Anthracene	secondary	5.68e-10	6.30e-11
Natural Gas Prod.	Cumene	secondary	5.50e-10	6.10e-11
Natural Gas Prod.	Indeno(1,2,3-cd)pyrene	secondary	5.26e-10	5.83e-11
Natural Gas Prod.	Acenaphthylene	secondary	4.89e-10	5.42e-11
Natural Gas Prod.	Benzo[b]fluoranthene	secondary	4.45e-10	4.93e-11
Natural Gas Prod.	Aromatic hydrocarbons	secondary	3.95e-10	4.39e-11
Natural Gas Prod.	Chrysene	secondary	3.79e-10	4.21e-11
Natural Gas Prod.	Biphenyl	secondary	3.28e-10	3.64e-11
Natural Gas Prod.	Benzo[g,h,i]perylene	secondary	2.23e-10	2.47e-11
Steel Prod., cold-rolled, semi-finished	Isopropylpropionate	secondary	1.98e-10	2.20e-11
Natural Gas Prod.	Copper (+1 & +2)	secondary	1.85e-10	2.05e-11
Natural Gas Prod.	Benzo[b,j,k]fluoranthene	secondary	1.78e-10	1.98e-11
Natural Gas Prod.	Pyrene	secondary	1.71e-10	1.90e-11
Natural Gas Prod.	Fluorene	secondary	1.14e-10	1.27e-11
Natural Gas Prod.	Fluoranthene	secondary	1.02e-10	1.13e-11
Steel Prod., cold-rolled, semi-finished	Lithium salts	secondary	9.71e-11	1.08e-11
Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
---	--	-----------------	-------------------------------------	------------
Steel Prod., cold-rolled, semi-finished	Perfluoromethane	secondary	4.40e-11	4.88e-12
Natural Gas Prod.	5-Methyl chrysene	secondary	3.57e-11	3.96e-12
Steel Prod., cold-rolled, semi-finished	Chloroform	secondary	2.69e-11	2.98e-12
Natural Gas Prod.	Lead cmpds	secondary	3.30e-12	3.66e-13
Natural Gas Prod.	Nickel cmpds	secondary	2.78e-12	3.09e-13
Natural Gas Prod.	HALON-1301	secondary	1.43e-12	1.59e-13
Natural Gas Prod.	Mercury compounds	secondary	1.00e-12	1.11e-13
Steel Prod., cold-rolled, semi-finished	Trichloroethylene	secondary	3.88e-13	4.30e-14
Natural Gas Prod.	Halogenated matter (organic)	secondary	3.29e-13	3.65e-14
Steel Prod., cold-rolled, semi-finished	Propionaldehyde	secondary	1.02e-13	1.13e-14
Steel Prod., cold-rolled, semi-finished	Benzaldehyde	secondary	3.01e-14	3.34e-15
Steel Prod., cold-rolled, semi-finished	Tetrachloroethylene	secondary	1.08e-14	1.20e-15
Steel Prod., cold-rolled, semi-finished	1,1,1-Trichloroethane	secondary	1.57e-15	1.75e-16
Natural Gas Prod.	Halogenated hydrocarbons (unspecified)	secondary	8.23e-16	9.13e-17
Steel Prod., cold-rolled, semi-finished	Hexachloroethane	secondary	6.03e-17	6.69e-18
Total Materials Processing			5.01e+01	5.56e+00
Manufacturing Life-cycle Stage				
Japanese Electric Grid	Sulfur dioxide	model/secondary	1.92e+02	2.12e+01
Monitor/module	Phosphine	primary	2.87e+01	3.18e+00
Monitor/module	Phosphorus (yellow or white)	primary	3.42e+00	3.79e-01
US electric grid	Sulfur dioxide	model/secondary	3.19e+00	3.54e-01
Monitor/module	Fluorides (F-)	primary	2.55e+00	2.83e-01
Monitor/module	Tetramethyl ammonium hydroxide	primary	1.29e+00	1.43e-01
Monitor/module	Nitrogen oxides	primary	1.10e+00	1.22e-01
Monitor/module	Nitrogen fluoride	primary	4.91e-01	5.44e-02
Monitor/module	Hydrochloric acid	primary	2.77e-01	3.08e-02
LPG Production	Carbon monoxide	secondary	2.77e-01	3.07e-02
Japanese Electric Grid	Nitrogen oxides	model/secondary	2.73e-01	3.03e-02
Monitor/module	Ammonia	primary	1.69e-01	1.88e-02
Natural Gas Prod.	Methane	secondary	1.58e-01	1.75e-02
Japanese Electric Grid	Carbon monoxide	model/secondary	1.27e-01	1.41e-02
Natural Gas Prod.	Benzene	secondary	1.19e-01	1.32e-02
Natural Gas Prod.	Carbon monoxide	secondary	1.08e-01	1.20e-02
Monitor/module	Hydrofluoric acid	primary	1.04e-01	1.16e-02
LPG Production	Methane	secondary	8.05e-02	8.93e-03
LPG Production	Sulfur oxides	secondary	7.69e-02	8.53e-03
Japanese Electric Grid	Vanadium	model/secondary	7.63e-02	8.46e-03
Backlight	Nitrogen oxides	primary	5.89e-02	6.53e-03
LPG Production	Nitrogen oxides	secondary	5.50e-02	6.10e-03
LPG Production	Vanadium	secondary	5.02e-02	5.57e-03
LPG Production	Benzene	secondary	4.12e-02	4.57e-03
Japanese Electric Grid	Arsenic	model/secondary	3.51e-02	3.89e-03
Natural Gas Prod.	Nitrogen oxides	secondary	2.94e-02	3.26e-03
LCD glass mfg.	Fluorides (F-)	primary	2.71e-02	3.01e-03
Monitor/module	Isopropyl alcohol	primary	2.23e-02	2.47e-03
Japanese Electric Grid	Hydrochloric acid	model/secondary	2.11e-02	2.34e-03
Panel components	Phosphorus (yellow or white)	primary	1.97e-02	2.19e-03
Monitor/module	N-bromoacetamide	primary	1.84e-02	2.04e-03
Monitor/module	Sulfur hexafluoride	primary	1.46e-02	1.62e-03
Japanese Electric Grid	PM-10	model/secondary	1.34e-02	1.49e-03

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
LPG Production	PM	secondary	1.24e-02	1.37e-03
LPG Production	Arsenic	secondary	9.91e-03	1.10e-03
LPG Production	Formaldehyde	secondary	6.43e-03	7.14e-04
US electric grid	Nitrogen oxides	model/secondary	4.56e-03	5.06e-04
Japanese Electric Grid	Fluorides (F-)	model/secondary	4.43e-03	4.91e-04
Japanese Electric Grid	Selenium	model/secondary	4.29e-03	4.76e-04
LCD glass mfg.	Nitrogen oxides	primary	4.09e-03	4.54e-04
Japanese Electric Grid	Formaldehyde	model/secondary	3.93e-03	4.36e-04
Monitor/module	Monosilane	primary	3.07e-03	3.41e-04
US electric grid	Methane	model/secondary	2.50e-03	2.77e-04
Fuel Oil #4 Prod.	Carbon monoxide	secondary	2.34e-03	2.59e-04
Monitor/module	Sulfur oxides	primary	2.23e-03	2.47e-04
US electric grid	Carbon monoxide	model/secondary	2.11e-03	2.34e-04
Japanese Electric Grid	Zinc (elemental)	model/secondary	2.06e-03	2.29e-04
PWB Mfg.	Formaldehyde	model/secondary	1.96e-03	2.18e-04
LPG Production	Hydrochloric acid	secondary	1.77e-03	1.96e-04
Monitor/module	Arsenic	primary	1.70e-03	1.88e-04
LPG Production	Nitrous oxide	secondary	1.56e-03	1.73e-04
Monitor/module	Acetic acid	primary	1.44e-03	1.60e-04
Natural Gas Prod	PM	secondary	1 31e-03	1 45e-04
Monitor/module	Hexane	primary	1.18e-03	1 31e-04
Japanese Electric Grid	Hydrofluoric acid	model/secondary	1.15e-03	1.28e-04
Japanese Electric Grid	Antimony	model/secondary	1.15e 03	1.20e 01
US electric grid	Arsenic	model/secondary	1.09e-03	1.20e-04
Fuel Oil #6 Prod	Carbon monoxide	secondary	9.64e-04	1.200 01
US electric grid	Hydrochloric acid	model/secondary	9.54e-04	1.06e-04
Natural Gas Prod	Sulfur ovides	secondary	9.54e 04 8.69e-04	9.64e-05
Panel components	Nitrogen oxides	primary	8.07e-04	9.12e-05
Fuel Oil #4 Prod	Methane	secondary	8.03e-04	9.12e-05
Fuel Oil #2 Prod	Carbon monovide	secondary	7 85e-04	8.71e-05
Fuel Oil #4 Prod	Sulfur oxides	secondary	7.55e-04	8 37e-05
Japanese Electric Grid	Nitrous oxide	model/secondary	7.536-04	8 34 9 05
Natural Gas Prod	Ammonia	secondary	6.71e-04	7.44e-05
Fuel Oil #4 Prod	Vanadium	secondary	6.21e.04	6 890 05
Fuel Oil #4 Prod	Vanaulum Nitrogen oxides	secondary	5 530 04	6 14e 05
Japanese Electric Grid	Methane	model/secondary	5.47e-04	6.07e-05
Monitor/module	Nitrie acid	nibuci/secondary	5.370.04	5.96e.05
Notural Gas Brad	Amonio	sacandam	4.060.04	5.500.05
L DG Production	Phosphorus (vallow or white)	secondary	4.900-04	5.306-03
Japanese Electric Grid	Malyhdanum	model/secondem	4./10-04	4.010.05
Fuel Oil #6 Pred	Mothana	model/secondary	4.430-04	4.916-03
Fuel Oil #6 Prod	Sulfar ovidor	secondary	2.72 . 04	4.450-05
Fuel Oil #6 Prod.	Sunur Oxides	secondary	3.72e-04	4.13e-03
Fuel Oil #6 Prod	Vanadium	secondary	2.67-04	4.120-05
L DC Droduction	vanautum Elverides (E.)	secondary	3.076-04	4.076-05
LPG Production	Fluorides (F-)	secondary	3.62e-04	4.01e-05
	Zing (alarmental)	secondary	3.566-04	3.94e-05
	Linc (elemental)	primary	3.13e-04	3.48e-05
Japanese Electric Grid	Benzene	model/secondary	3.06e-04	3.40e-05
LPG Production	Hydrogen sulfide	secondary	2.94e-04	3.26e-05

 Table M-34. LCD LCIA Results for the Chronic Public Health Effects Impact Category

Toxicity (tox-kg)LPG ProductionAmmoniasecondary2.84e-043.15e-05Panel componentsHCFC-225caprimary2.80e-043.10e-05Panel componentsHCFC-225cbprimary2.80e-043.10e-05Fuel Oil #6 Prod.Nitrogen oxidessecondary2.79e-043.09e-05Fuel Oil #2 Prod.Methanesecondary2.40e-042.66e-05Monitor/modulePhosphoric acidprimary2.33e-042.59e-05Fuel Oil #2 Prod.Sulfur oxidessecondary2.27e-042.52e-05US electric gridPM-10model/secondary2.24e-042.48e-05US electric gridSeleniummodel/secondary1.80e-041.99e-05
LPG ProductionAmmoniasecondary2.84e-043.15e-05Panel componentsHCFC-225caprimary2.80e-043.10e-05Panel componentsHCFC-225cbprimary2.80e-043.10e-05Fuel Oil #6 Prod.Nitrogen oxidessecondary2.79e-043.09e-05Fuel Oil #2 Prod.Methanesecondary2.40e-042.66e-05Monitor/modulePhosphoric acidprimary2.33e-042.59e-05Fuel Oil #2 Prod.Sulfur oxidessecondary2.27e-042.52e-05US electric gridPM-10model/secondary2.24e-042.48e-05US electric gridSeleniummodel/secondary1.80e-041.99e-05
Panel componentsHCFC-225caprimary2.80e-043.10e-05Panel componentsHCFC-225cbprimary2.80e-043.10e-05Fuel Oil #6 Prod.Nitrogen oxidessecondary2.79e-043.09e-05Fuel Oil #2 Prod.Methanesecondary2.40e-042.66e-05Monitor/modulePhosphoric acidprimary2.33e-042.59e-05Fuel Oil #2 Prod.Sulfur oxidessecondary2.27e-042.52e-05US electric gridPM-10model/secondary2.24e-042.48e-05US electric gridSeleniummodel/secondary1.80e-041.99e-05
Panel componentsHCFC-225cbprimary2.80e-043.10e-05Fuel Oil #6 Prod.Nitrogen oxidessecondary2.79e-043.09e-05Fuel Oil #2 Prod.Methanesecondary2.40e-042.66e-05Monitor/modulePhosphoric acidprimary2.33e-042.59e-05Fuel Oil #2 Prod.Sulfur oxidessecondary2.27e-042.52e-05US electric gridPM-10model/secondary2.24e-042.48e-05US electric gridSeleniummodel/secondary1.80e-041.99e-05
Fuel Oil #6 Prod.Nitrogen oxidessecondary2.79e-043.09e-05Fuel Oil #2 Prod.Methanesecondary2.40e-042.66e-05Monitor/modulePhosphoric acidprimary2.33e-042.59e-05Fuel Oil #2 Prod.Sulfur oxidessecondary2.27e-042.52e-05US electric gridPM-10model/secondary2.24e-042.48e-05US electric gridSeleniummodel/secondary1.80e-041.99e-05
Fuel Oil #2 Prod.Methanesecondary2.40e-042.66e-05Monitor/modulePhosphoric acidprimary2.33e-042.59e-05Fuel Oil #2 Prod.Sulfur oxidessecondary2.27e-042.52e-05US electric gridPM-10model/secondary2.24e-042.48e-05US electric gridSeleniummodel/secondary1.80e-041.99e-05
Monitor/modulePhosphoric acidprimary2.33e-042.59e-05Fuel Oil #2 Prod.Sulfur oxidessecondary2.27e-042.52e-05US electric gridPM-10model/secondary2.24e-042.48e-05US electric gridSeleniummodel/secondary1.80e-041.99e-05
Fuel Oil #2 Prod.Sulfur oxidessecondary2.27e-042.52e-05US electric gridPM-10model/secondary2.24e-042.48e-05US electric gridSeleniummodel/secondary1.80e-041.99e-05
US electric gridPM-10model/secondary2.24e-042.48e-05US electric gridSeleniummodel/secondary1.80e-041.99e-05
US electric grid Selenium model/secondary 1.80e-04 1.99e-05
Japanese Electric GridNickelmodel/secondary1.67e-041.86e-05
Fuel Oil #6 Prod.Benzenesecondary1.67e-041.85e-05
Fuel Oil #2 Prod.Nitrogen oxidessecondary1.64e-041.82e-05
Fuel Oil #2 Prod.Vanadiumsecondary1.61e-041.79e-05
US electric grid Vanadium model/secondary 1.35e-04 1.50e-05
Fuel Oil #4 Prod. PM secondary 1.25e-04 1.38e-05
Fuel Oil #2 Prod.Benzenesecondary1.19e-041.32e-05
Natural Gas Prod. Vanadium secondary 1.10e-04 1.22e-05
Fuel Oil #4 Prod.Arsenicsecondary1.09e-041.20e-05
Natural Gas Prod.Hydrochloric acidsecondary1.01e-041.12e-05
Japanese Electric Grid Barium model/secondary 9.88e-05 1.10e-05
Monitor/module Diethylene glycol primary 9.79e-05 1.09e-05
LPG Production Hydrofluoric acid secondary 9.63e-05 1.07e-05
BacklightDiethyl etherprimary9.49e-051.05e-05
Fuel Oil #4 Prod.Formaldehydesecondary7.92e-058.78e-06
LPG ProductionMolybdenumsecondary7.91e-058.77e-06
LPG Production Chromium (VI) secondary 7.57e-05 8.39e-06
LPG Production Ethane secondary 7.44e-05 8.26e-06
Fuel Oil #6 Prod. PM secondary 6.29e-05 6.98e-06
Fuel Oil #6 Prod.Arsenicsecondary5.87e-056.51e-06
LPG ProductionZinc (elemental)secondary5.75e-056.38e-06
Japanese Electric GridNaphthalenemodel/secondary5.57e-056.18e-06
Panel components PM primary 5.48e-05 6.08e-06
US electric grid Hydrofluoric acid model/secondary 5.21e-05 5.78e-06
Fuel Oil #6 Prod.Formaldehydesecondary4.66e-055.17e-06
Natural Gas Prod.Formaldehydesecondary4.37e-054.85e-06
LPG Production Hexane secondary 4.33e-05 4.80e-06
Monitor/module Antimony primary 3.89e-05 4.32e-06
Fuel Oil #2 Prod. PM secondary 3.69e-05 4.10e-06
Natural Gas Prod.Nitrous oxidesecondary3.64e-054.04e-06
LPG Production Phenol secondary 3.43e-05 3.81e-06
Japanese Electric Grid Chromium (VI) model/secondary 3.36e-05 3.73e-06
Panel components Hydrochloric acid primary 3.35e-05 3.72e-06
Panel components Heptane primary 3.28e-05 3.63e-06
LPG Production Pentane secondary 3.12e-05 3.46e-06
Fuel Oil #2 Prod.Arsenicsecondary3.04e-053.38e-06
Japanese Electric Grid Copper model/secondary 2.40e-05 2.67e-06
Monitor/module Acetone primary 2.22e-05 2.46e-06
Natural Gas Prod. Ethane secondary 2.20e-05 2.44e-06
Monitor/module PM primary 2.20e-05 2.44e-06
LPG Production PM-10 secondary 2.16e-05 2.40e-06
Natural Gas Prod.Fluorides (F-)secondary2.14e-052.38e-06

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Monitor/module	Copper	primary	2.06e-05	2.29e-06
Fuel Oil #2 Prod.	Formaldehyde	secondary	2.06e-05	2.28e-06
LPG Production	Nickel	secondary	1.98e-05	2.19e-06
Japanese Electric Grid	Methyl hydrazine	model/secondary	1.98e-05	2.19e-06
Monitor/module	Polychlorinated biphenyls	primary	1.94e-05	2.15e-06
Natural Gas Prod.	Selenium	secondary	1.91e-05	2.12e-06
Fuel Oil #4 Prod.	Hydrochloric acid	secondary	1.90e-05	2.10e-06
US electric grid	Formaldehyde	model/secondary	1.88e-05	2.08e-06
Japanese Electric Grid	2-Chloroacetophenone	model/secondary	1.85e-05	2.05e-06
Japanese Electric Grid	Bromomethane	model/secondary	1.83e-05	2.03e-06
Fuel Oil #4 Prod.	Nitrous oxide	secondary	1.77e-05	1.97e-06
Monitor/module	Chromium	primary	1.77e-05	1.96e-06
Monitor/module	Chromium (VI)	primary	1.71e-05	1.89e-06
LPG Production	Aluminum (+3)	secondary	1.36e-05	1.51e-06
US electric grid	Benzene	model/secondary	1.35e-05	1.50e-06
Natural Gas Prod.	Zinc (elemental)	secondary	1.34e-05	1.49e-06
US electric grid	Nitrous oxide	model/secondary	1.32e-05	1.46e-06
Natural Gas Prod.	Hexane	secondary	1.28e-05	1.42e-06
Monitor/module	Lead	primary	1.23e-05	1.37e-06
Monitor/module	Boron	primary	1.08e-05	1.20e-06
LPG Production	Antimony	secondary	1.03e-05	1.14e-06
LCD glass mfg.	PM	primary	1.01e-05	1.12e-06
Fuel Oil #6 Prod.	Hydrochloric acid	secondary	1.01e-05	1.12e-06
Fuel Oil #6 Prod.	Nitrous oxide	secondary	9.88e-06	1.10e-06
Japanese Electric Grid	Cyanide (-1)	model/secondary	9.61e-06	1.07e-06
Natural Gas Prod.	Pentane	secondary	9.24e-06	1.02e-06
Panel components	Toluene	primary	9.09e-06	1.01e-06
Japanese Electric Grid	Cobalt	model/secondary	7.96e-06	8.83e-07
US electric grid	Phosphorus (vellow or white)	model/secondary	7.61e-06	8.44e-07
Japanese Electric Grid	2.3.7.8-TCDD	model/secondary	7.49e-06	8.30e-07
LPG Production	Nitrate	secondary	5.81e-06	6.45e-07
Fuel Oil #4 Prod.	Phosphorus (vellow or white)	secondary	5.71e-06	6.33e-07
Natural Gas Prod.	Hydrofluoric acid	secondary	5.49e-06	6.09e-07
Japanese Electric Grid	Cadmium	model/secondary	5.39e-06	5.98e-07
Fuel Oil #2 Prod.	Hydrochloric acid	secondary	5.39e-06	5.98e-07
Fuel Oil #2 Prod.	Nitrous oxide	secondary	4.84e-06	5.37e-07
LCD glass mfg.	Sulfur oxides	primary	4.71e-06	5.22e-07
LCD glass mfg	Lead	primary	4 02e-06	4 46e-07
Japanese Electric Grid	Carbon disulfide	model/secondary	3 93e-06	4 36e-07
Fuel Oil #4 Prod	Selenium	secondary	3.85e-06	4 27e-07
Fuel Oil #4 Prod	Fluorides (F-)	secondary	3.81e-06	4 23e-07
Monitor/module	Cvanide (-1)	primary	3.66e-06	4.06e-07
Fuel Oil #4 Prod	Hydrogen sulfide	secondary	3.65e-06	4.00e 07
Natural Gas Prod	Molyhdenum	secondary	3.65e-06	4.05e-07
US electric grid	Zinc (elemental)	model/secondary	3 510 06	3.800-07
Fuel Oil #6 Prod	Phosphorus (vellow or white)	secondary	3 330.06	3.69e-07
Japanese Electric Grid	Benzyl chloride	model/secondary	3.010.06	3 340 07
US electric grid	Antimony	model/secondary	2 880 06	3.200.07
I CD glass mfg	Carbon monovide	nrimory	2.000-00	3 100 07
LCD glass ling.	Carbon monoriae	Printary	2.798-00	5.100-07

 Table M-34.
 LCD LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Phosphorus (yellow or white)	secondary	2.79e-06	3.09e-07
Monitor/module	Hexamethyldisilizane	primary	2.75e-06	3.04e-07
Fuel Oil #4 Prod.	Ammonia	secondary	2.49e-06	2.77e-07
Natural Gas Prod.	Chromium (VI)	secondary	2.33e-06	2.59e-07
Monitor/module	Cyclohexane	primary	2.22e-06	2.46e-07
Fuel Oil #6 Prod.	Hydrogen sulfide	secondary	2.16e-06	2.40e-07
Japanese Electric Grid	Chloroform	model/secondary	2.12e-06	2.36e-07
Fuel Oil #6 Prod.	Selenium	secondary	2.07e-06	2.29e-07
Fuel Oil #6 Prod.	Fluorides (F-)	secondary	2.00e-06	2.21e-07
Japanese Electric Grid	Propionaldehyde	model/secondary	1.96e-06	2.18e-07
LPG Production	Copper	secondary	1.67e-06	1.86e-07
Japanese Electric Grid	Beryllium	model/secondary	1.66e-06	1.84e-07
LPG Production	Methyl hydrazine	secondary	1.65e-06	1.83e-07
LPG Production	Silicon	secondary	1.62e-06	1.80e-07
LPG Production	2-Chloroacetophenone	secondary	1.55e-06	1.72e-07
LPG Production	Dimethylbenzanthracene	secondary	1.53e-06	1.70e-07
LPG Production	Bromomethane	secondary	1.53e-06	1.70e-07
Fuel Oil #2 Prod.	Phosphorus (yellow or white)	secondary	1.50e-06	1.66e-07
Monitor/module	Cadmium	primary	1.30e-06	1.44e-07
LPG Production	Naphthalene	secondary	1.25e-06	1.39e-07
LPG Production	Lead	secondary	1.19e-06	1.32e-07
Japanese Electric Grid	Acrolein	model/secondary	1.11e-06	1.24e-07
Fuel Oil #2 Prod.	Fluorides (F-)	secondary	1.10e-06	1.22e-07
Fuel Oil #2 Prod.	Selenium	secondary	1.09e-06	1.21e-07
Fuel Oil #6 Prod.	Ammonia	secondary	1.08e-06	1.20e-07
LPG Production	Manganese	secondary	1.07e-06	1.19e-07
US electric grid	Chromium (VI)	model/secondary	1.04e-06	1.15e-07
Fuel Oil #4 Prod.	Hydrofluoric acid	secondary	1.03e-06	1.15e-07
LPG Production	Beryllium	secondary	1.03e-06	1.14e-07
US electric grid	Molybdenum	model/secondary	9.62e-07	1.07e-07
Fuel Oil #2 Prod.	Hydrogen sulfide	secondary	9.44e-07	1.05e-07
Fuel Oil #4 Prod.	Molybdenum	secondary	9.30e-07	1.03e-07
Japanese Electric Grid	Methyl chloride	model/secondary	9.25e-07	1.03e-07
Monitor/module	Tin	primary	9.16e-07	1.02e-07
LPG Production	Barium	secondary	8.94e-07	9.92e-08
US electric grid	Methyl hydrazine	model/secondary	8.93e-07	9.90e-08
US electric grid	2-Chloroacetophenone	model/secondary	8.36e-07	9.28e-08
US electric grid	Bromomethane	model/secondary	8.27e-07	9.17e-08
Fuel Oil #2 Prod.	Ammonia	secondary	8.14e-07	9.03e-08
Japanese Electric Grid	Toluene	model/secondary	8.04e-07	8.92e-08
LPG Production	Cvanide (-1)	secondary	8 03e-07	8 90e-08
Monitor/module	Nickel	nrimary	7 74e-07	8 59e-08
LPG Production	Barium cmpds	secondary	7.62e-07	8 45e-08
Natural Gas Prod	Antimony	secondary	7.02e 07	7 79e-08
Fuel Oil #4 Prod	Fthane	secondary	6 72e-07	7.45e-08
Japanese Electric Grid	Di(2-ethylhexyl)phthalate	model/secondary	6.66e-07	7 39e-08
Fuel Oil #6 Prod	Hydrofluoric acid	secondary	5 52e-07	6 12e-08
Fuel Oil #6 Prod	Molyhdenum	secondary	5 31e 07	5.800-08
Jananese Electric Grid	Acetaldehyde	model/secondary	5 31e-07	5 88e-08
Japanese Electric Grid	Hevane	model/secondary	5 15e 07	5.710-08
Fuel Oil #4 Prod	Zinc (elemental)	secondary	5.100.07	5.66- 09
$1 u c 1 0 11 \pi + 1 10 u.$	Zine (elemental)	secondary	5.108-07	5.000-08

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Dimethylbenzanthracene	secondary	4.53e-07	5.03e-08
US electric grid	Nickel	model/secondary	4.48e-07	4.97e-08
LPG Production	Cadmium	secondary	4.42e-07	4.90e-08
US electric grid	Cyanide (-1)	model/secondary	4.34e-07	4.82e-08
Fuel Oil #4 Prod.	Chromium (VI)	secondary	4.07e-07	4.52e-08
Japanese Electric Grid	Mercury	model/secondary	3.97e-07	4.41e-08
Fuel Oil #4 Prod.	Hexane	secondary	3.91e-07	4.33e-08
Japanese Electric Grid	Dimethyl sulfate	model/secondary	3.69e-07	4.09e-08
LCD glass mfg.	Nitrate	primary	3.66e-07	4.06e-08
US electric grid	2,3,7,8-TCDD	model/secondary	3.29e-07	3.65e-08
LPG Production	Carbon disulfide	secondary	3.28e-07	3.64e-08
Fuel Oil #6 Prod.	Ethane	secondary	3.02e-07	3.35e-08
Fuel Oil #2 Prod.	Hydrofluoric acid	secondary	2.94e-07	3.26e-08
Fuel Oil #4 Prod.	Pentane	secondary	2.82e-07	3.13e-08
Natural Gas Prod.	Nickel	secondary	2.75e-07	3.05e-08
Fuel Oil #4 Prod.	PM-10	secondary	2.68e-07	2.97e-08
US electric grid	Naphthalene	model/secondary	2.65e-07	2.94e-08
LPG Production	Benzyl chloride	secondary	2.51e-07	2.79e-08
Fuel Oil #4 Prod.	Phenol	secondary	2.51e-07	2.79e-08
Fuel Oil #2 Prod.	Molybdenum	secondary	2.49e-07	2.76e-08
LPG Production	Cobalt	secondary	2.41e-07	2.67e-08
Fuel Oil #4 Prod.	Nickel	secondary	2.36e-07	2.61e-08
Monitor/module	Manganese	primary	2.29e-07	2.54e-08
US electric grid	Barium	model/secondary	2.25e-07	2.50e-08
Fuel Oil #6 Prod.	Zinc (elemental)	secondary	2.25e-07	2.49e-08
Fuel Oil #6 Prod.	Chromium (VI)	secondary	2.18e-07	2.41e-08
Fuel Oil #2 Prod.	Ethane	secondary	2.15e-07	2.39e-08
Natural Gas Prod.	Hydrogen sulfide	secondary	2.12e-07	2.35e-08
Natural Gas Prod.	Naphthalene	secondary	2.08e-07	2.31e-08
LPG Production	Aluminum (elemental)	secondary	1.86e-07	2.06e-08
Japanese Electric Grid	Isophorone	model/secondary	1 78e-07	1 98e-08
US electric grid	Carbon disulfide	model/secondary	1 78e-07	1 97e-08
LPG Production	Chloroform	secondary	1.77e-07	1.97e-08
Fuel Oil #6 Prod	Hexane	secondary	1 76e-07	1 95e-08
Fuel Oil #2 Prod	Zinc (elemental)	secondary	1.65e-07	1.84e-08
LPG Production	Propionaldehyde	secondary	1.64e-07	1.82e-08
Fuel Oil #6 Prod	PM-10	secondary	1.59e-07	1 76e-08
Japanese Electric Grid	Methyl ethyl ketone	model/secondary	1.39e-07	1.58e-08
Japanese Electric Grid	Tetrachloroethylene	model/secondary	1.13e 07	1.56e-08
US electric grid	Benzyl chloride	model/secondary	1.41e 07	1.50e-08
Fuel Oil #6 Prod	Nickel	secondary	1.36e-07	1.51e-08
Fuel Oil #6 Prod	Pentane	secondary	1.30e 07	1.51e 00
Japanese Electric Grid	1 2-Dichloroethane	model/secondary	1.27e-07	1.40c-08
Fuel Oil #2 Prod	Hevane	secondary	1.25e-07	1.39e-08
Japanese Electric Grid	Methyl methacrylate	model/secondary	1.230-07	1.350-08
Fuel Oil #2 Prod	Chromium (VI)	secondary	1.220-07	1.330-08
I uci Oli #2 1100. Japanese Electric Grid	Styrene	model/secondary	1.100-07	1.200-08
US electric grid	Cadmium	model/secondary	1.0607	1.100-08
Japanese Electric Grid	Bromoform	model/secondary	1.000-07	1.100-08
Japanese Elecule Offu	DIOMOIOIIII	mouch secondary	1.016-07	1.120-08

 Table M-34.
 LCD LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Fuel Oil #4 Prod.	Aluminum (+3)	secondary	9.95e-08	1.10e-08
Japanese Electric Grid	Dichloromethane	model/secondary	9.94e-08	1.10e-08
Monitor/module	Mercury	primary	9.69e-08	1.08e-08
US electric grid	Chloroform	model/secondary	9.59e-08	1.06e-08
Natural Gas Prod.	Methyl hydrazine	secondary	9.41e-08	1.04e-08
Fuel Oil #2 Prod.	Phenol	secondary	9.37e-08	1.04e-08
LPG Production	Acrolein	secondary	9.31e-08	1.03e-08
Natural Gas Prod.	Barium	secondary	9.30e-08	1.03e-08
Fuel Oil #4 Prod.	Antimony	secondary	9.15e-08	1.02e-08
Fuel Oil #2 Prod.	Pentane	secondary	9.02e-08	1.00e-08
US electric grid	Propionaldehyde	model/secondary	8.86e-08	9.83e-09
US electric grid	Manganese	model/secondary	8.83e-08	9.79e-09
Natural Gas Prod.	2-Chloroacetophenone	secondary	8.81e-08	9.77e-09
Natural Gas Prod.	Bromomethane	secondary	8.71e-08	9.66e-09
Natural Gas Prod.	Copper	secondary	8.32e-08	9.23e-09
Fuel Oil #6 Prod.	Phenol	secondary	8.24e-08	9.14e-09
Japanese Electric Grid	Chlorobenzene	model/secondary	8.05e-08	8.93e-09
LPG Production	Methyl chloride	secondary	7.72e-08	8.57e-09
US electric grid	Fluoride	model/secondary	7.56e-08	8.39e-09
Panel components	Methyl ethyl ketone	primary	6.97e-08	7.73e-09
Fuel Oil #2 Prod.	PM-10	secondary	6.93e-08	7.69e-09
Natural Gas Prod.	Lead	secondary	6.83e-08	7.58e-09
Natural Gas Prod.	Beryllium	secondary	6.74e-08	7.47e-09
Japanese Electric Grid	Cumene hydroperoxide	model/secondary	6.55e-08	7.27e-09
Japanese Electric Grid	2,4-Dinitrotoluene	model/secondary	6.44e-08	7.14e-09
Fuel Oil #2 Prod.	Nickel	secondary	6.25e-08	6.94e-09
Natural Gas Prod.	Manganese	secondary	6.23e-08	6.91e-09
Japanese Electric Grid	Acetophenone	model/secondary	5.93e-08	6.58e-09
US electric grid	Beryllium	model/secondary	5.64e-08	6.26e-09
LPG Production	Di(2-ethylhexyl)phthalate	secondary	5.56e-08	6.17e-09
US electric grid	Acrolein	model/secondary	5.04e-08	5.59e-09
US electric grid	Lead	model/secondary	4.94e-08	5.48e-09
Natural Gas Prod.	Nitrate	secondary	4.74e-08	5.26e-09
US electric grid	Cobalt	model/secondary	4.71e-08	5.22e-09
Natural Gas Prod.	Cvanide (-1)	secondary	4.57e-08	5.07e-09
Monitor/module	Phenol	primary	4.54e-08	5.04e-09
Fuel Oil #4 Prod.	Nitrate	secondary	4.52e-08	5.02e-09
LPG Production	Acetaldehvde	secondary	4.43e-08	4.91e-09
US electric grid	Copper	model/secondary	4.33e-08	4.80e-09
US electric grid	Methyl chloride	model/secondary	4.18e-08	4.63e-09
LPG Production	Mercury	secondary	4 17e-08	4 62e-09
Fuel Oil #6 Prod	Antimony	secondary	4 05e-08	4 49e-09
Fuel Oil #2 Prod	Aluminum (+3)	secondary	3 71e-08	4 11e-09
Jananese Electric Grid	Fthylbenzene	model/secondary	3.49e-08	3 88e-09
LPG Production	Cadmium cmpds	secondary	3.33e-08	3.70e-09
Fuel Oil #6 Prod	Aluminum (+3)	secondary	3.26e-08	3.62e-09
LPG Production	Dimethyl sulfate	secondary	3 080-08	3 470-09
US electric grid	Di(2-ethylbeyyl)nhthalate	model/secondary	3.010.08	3.340-00
LPG Production	Toluene	secondary	2 000.08	3 370-09
Fuel Oil #2 Prod	Antimony	secondary	2.550-08	3.28-09
Natural Gas Prod	Cadmium	secondary	2.500-00	2 70- 00
ivatural Gas 1100.	Caumum	secondary	2.528-08	2.790-09

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
US electric grid	Acetaldehyde	model/secondary	2.40e-08	2.66e-09
US electric grid	Hexane	model/secondary	2.33e-08	2.58e-09
Natural Gas Prod.	Phenol	secondary	2.30e-08	2.55e-09
LCD glass mfg.	Chromium	primary	2.04e-08	2.26e-09
Fuel Oil #4 Prod.	Silicon	secondary	2.02e-08	2.24e-09
Monitor/module	Tetrachloroethylene	primary	1.95e-08	2.17e-09
Japanese Electric Grid	Acenaphthene	model/secondary	1.95e-08	2.16e-09
Fuel Oil #4 Prod.	Copper	secondary	1.92e-08	2.13e-09
Natural Gas Prod.	Carbon disulfide	secondary	1.87e-08	2.08e-09
Fuel Oil #4 Prod.	Methyl hydrazine	secondary	1.77e-08	1.97e-09
Japanese Electric Grid	Phenanthrene	model/secondary	1.75e-08	1.94e-09
LPG Production	1,4-Dichlorobenzene	secondary	1.74e-08	1.93e-09
US electric grid	Dimethyl sulfate	model/secondary	1.67e-08	1.85e-09
Fuel Oil #4 Prod.	2-Chloroacetophenone	secondary	1.66e-08	1.84e-09
Fuel Oil #6 Prod.	Nitrate	secondary	1.65e-08	1.83e-09
Fuel Oil #4 Prod.	Bromomethane	secondary	1.64e-08	1.82e-09
Fuel Oil #2 Prod.	Nitrate	secondary	1.61e-08	1.79e-09
Japanese Electric Grid	Methyl tert-butyl ether	model/secondary	1.60e-08	1.77e-09
Natural Gas Prod.	PM-10	secondary	1.56e-08	1.73e-09
Japanese Electric Grid	Chromium (III)	model/secondary	1.51e-08	1.67e-09
LPG Production	Isophorone	secondary	1.49e-08	1.65e-09
US electric grid	Mercury	model/secondary	1.45e-08	1.61e-09
Natural Gas Prod.	Benzyl chloride	secondary	1.43e-08	1.59e-09
Fuel Oil #4 Prod.	Dimethylbenzanthracene	secondary	1.38e-08	1.53e-09
Natural Gas Prod.	Toluene	secondary	1.36e-08	1.51e-09
LPG Production	Aromatic hydrocarbons	secondary	1.34e-08	1.49e-09
LCD glass mfg.	Nickel	primary	1.28e-08	1.42e-09
Fuel Oil #4 Prod.	Lead	secondary	1.28e-08	1.41e-09
Japanese Electric Grid	1.1.1-Trichloroethane	model/secondary	1.23e-08	1.36e-09
Japanese Electric Grid	Phenol	model/secondary	1.22e-08	1.35e-09
Fuel Oil #6 Prod.	Silicon	secondary	1.19e-08	1.32e-09
LPG Production	Methyl ethyl ketone	secondary	1 19e-08	1 32e-09
LPG Production	Tetrachloroethylene	secondary	1.18e-08	1.31e-09
Fuel Oil #4 Prod	Naphthalene	secondary	1 16e-08	1 29e-09
Monitor/module	Trichloroethylene	primary	1 16e-08	1 29e-09
Fuel Oil #4 Prod.	Manganese	secondary	1.15e-08	1.27e-09
Japanese Electric Grid	Vinvl acetate	model/secondary	1 14e-08	1 26e-09
Fuel Oil #4 Prod	Beryllium	secondary	1 10e-08	1.20e-09
Fuel Oil #6 Prod	Copper	secondary	1.10e 00	1.22e 09
Japanese Electric Grid	Ethylene dibromide	model/secondary	1.00 c -08	1.20e 09
LPG Production	1.2-Dichloroethane	secondary	1.00e-08	1.1/e 09
LPG Production	Methyl methacrylate	secondary	1.02e-08	1.13e-09
Natural Gas Prod	Chloroform	secondary	1.02e 08	1.13e-09
Jananese Electric Grid	Xylene (mixed isomers)	model/secondary	9.46e-09	1.05e-09
Fuel Oil #6 Prod	Methyl hydrazine	secondary	9.450.09	1.05e-09
Natural Gas Prod	Pronionaldehyde	secondary	0.3/2.00	1.040.00
Natural Gas Prod	Aluminum (+3)	secondary	0.030.00	1.040-09
I PG Production	Styrene	secondary	9.030-09	0.00-10
Fuel Oil #6 Prod	2-Chlorogestonhenone	secondary	9.000-09 8.85 ₀ .00	9.990-10
	2-Chloroacetophenone	secondary	0.050-09	7.620-10

Table M-34. LCD LCIA Results for the Chronic Public Health Effects Impact Category

Process Crown	Matarial	LCI Data Tuna	Chronic Public	V of Total
Process Group	Material	LCI Data Type	Toxicity (tox-kg)	70 01 10tai
Fuel Oil #6 Prod.	Bromomethane	secondary	8.75e-09	9.71e-10
Fuel Oil #4 Prod.	Cyanide (-1)	secondary	8.62e-09	9.56e-10
LPG Production	Bromoform	secondary	8.41e-09	9.33e-10
Natural Gas Prod.	Cobalt	secondary	8.33e-09	9.24e-10
LPG Production	Dichloromethane	secondary	8.30e-09	9.20e-10
LPG Production	Chromium (III)	secondary	8.24e-09	9.14e-10
US electric grid	Toluene	model/secondary	8.23e-09	9.12e-10
US electric grid	Isophorone	model/secondary	8.06e-09	8.94e-10
Fuel Oil #4 Prod.	Barium	secondary	7.83e-09	8.69e-10
Natural Gas Prod.	o-xylene	secondary	7.25e-09	8.04e-10
Fuel Oil #6 Prod.	Lead	secondary	6.80e-09	7.55e-10
LPG Production	Chlorobenzene	secondary	6.72e-09	7.46e-10
US electric grid	Methyl ethyl ketone	model/secondary	6.45e-09	7.15e-10
US electric grid	Tetrachloroethylene	model/secondary	6.37e-09	7.07e-10
LPG Production	Copper (+1 & +2)	secondary	6.29e-09	6.97e-10
Japanese Electric Grid	Dibenzo[a,h]anthracene	model/secondary	6.24e-09	6.92e-10
Fuel Oil #6 Prod.	Dimethylbenzanthracene	secondary	6.21e-09	6.89e-10
Fuel Oil #6 Prod.	Manganese	secondary	6.09e-09	6.76e-10
Fuel Oil #6 Prod.	Beryllium	secondary	5.82e-09	6.46e-10
US electric grid	1,2-Dichloroethane	model/secondary	5.67e-09	6.29e-10
Fuel Oil #4 Prod.	Barium cmpds	secondary	5.58e-09	6.19e-10
US electric grid	Methyl methacrylate	model/secondary	5.51e-09	6.11e-10
LPG Production	2,4-Dinitrotoluene	secondary	5.39e-09	5.98e-10
Fuel Oil #6 Prod.	Naphthalene	secondary	5.39e-09	5.98e-10
Natural Gas Prod.	Acrolein	secondary	5.31e-09	5.89e-10
Fuel Oil #2 Prod.	Copper	secondary	5.22e-09	5.79e-10
Fuel Oil #2 Prod.	Silicon	secondary	5.21e-09	5.78e-10
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	5.15e-09	5.71e-10
Fuel Oil #2 Prod.	Methyl hydrazine	secondary	5.04e-09	5.59e-10
Japanese Electric Grid	2,3,7,8-TCDF	model/secondary	5.01e-09	5.56e-10
LPG Production	Acetophenone	secondary	4.95e-09	5.49e-10
US electric grid	Styrene	model/secondary	4.87e-09	5.40e-10
Fuel Oil #2 Prod.	2-Chloroacetophenone	secondary	4.72e-09	5.23e-10
Fuel Oil #4 Prod.	Cadmium	secondary	4.68e-09	5.19e-10
Fuel Oil #2 Prod.	Bromomethane	secondary	4.66e-09	5.17e-10
Fuel Oil #6 Prod.	Cyanide (-1)	secondary	4.60e-09	5.10e-10
US electric grid	Bromoform	model/secondary	4.55e-09	5.04e-10
US electric grid	Dichloromethane	model/secondary	4.49e-09	4.98e-10
Fuel Oil #2 Prod.	Dimethylbenzanthracene	secondary	4.43e-09	4.91e-10
Natural Gas Prod.	Methyl chloride	secondary	4.40e-09	4.88e-10
Japanese Electric Grid	o-xylene	model/secondary	4 31e-09	4 79e-10
Japanese Electric Grid	Benzo[a]anthracene	model/secondary	4 11e-09	4 56e-10
Fuel Oil #2 Prod	Naphthalene	secondary	3 65e-09	4 05e-10
US electric grid	Chlorobenzene	model/secondary	3 64e-09	4 03e-10
Fuel Oil #2 Prod	Lead	secondary	3.62e-09	4 02e-10
Fuel Oil #4 Prod	Carbon disulfide	secondary	3 53e-09	3.91e-10
Fuel Oil #6 Prod	Barium	secondary	3 39e-09	3 76e-10
LPG Production	o-xylene	secondary	3 29e-09	3.65e-10
Fuel Oil #2 Prod	Manganese	secondary	3.250-09	3.62e-10
Natural Gas Prod	Di(2-ethylbexyl)phthalate	secondary	3.17e-09	3 52e-10
Fuel Oil #2 Prod	Bervllium	secondary	3 13e-00	3 480-10
		secondary	5.150-09	5.400-10

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Japanese Electric Grid	Ethyl Chloride	model/secondary	3.08e-09	3.42e-10
Natural Gas Prod.	Mercury	secondary	2.97e-09	3.29e-10
US electric grid	2,4-Dinitrotoluene	model/secondary	2.92e-09	3.23e-10
Fuel Oil #4 Prod.	Benzyl chloride	secondary	2.70e-09	3.00e-10
US electric grid	Acetophenone	model/secondary	2.68e-09	2.97e-10
LPG Production	Ethylbenzene	secondary	2.67e-09	2.96e-10
Fuel Oil #4 Prod.	Cobalt	secondary	2.66e-09	2.95e-10
Japanese Electric Grid	Benzo[b,j,k]fluoranthene	model/secondary	2.61e-09	2.90e-10
Fuel Oil #2 Prod.	Barium	secondary	2.56e-09	2.84e-10
Natural Gas Prod.	Acetaldehyde	secondary	2.52e-09	2.80e-10
Fuel Oil #6 Prod.	Cadmium	secondary	2.46e-09	2.73e-10
Fuel Oil #2 Prod.	Cyanide (-1)	secondary	2.45e-09	2.72e-10
Fuel Oil #4 Prod.	Aluminum (elemental)	secondary	2.31e-09	2.56e-10
Japanese Electric Grid	Indeno(1,2,3-cd)pyrene	model/secondary	2.30e-09	2.55e-10
Fuel Oil #2 Prod.	Barium cmpds	secondary	2.08e-09	2.31e-10
Fuel Oil #4 Prod.	Chloroform	secondary	1.91e-09	2.11e-10
Fuel Oil #6 Prod.	Carbon disulfide	secondary	1.88e-09	2.09e-10
Fuel Oil #6 Prod.	Barium cmpds	secondary	1.83e-09	2.03e-10
Japanese Electric Grid	Chrysene	model/secondary	1.81e-09	2.01e-10
Fuel Oil #4 Prod.	Propionaldehvde	secondary	1.76e-09	1.95e-10
Natural Gas Prod.	Dimethyl sulfate	secondary	1.76e-09	1.95e-10
Natural Gas Prod.	Z_{inc} (+2)	secondary	1.65e-09	1.83e-10
Japanese Electric Grid	2-Methylnanhthalene	model/secondary	1 60e-09	1 77e-10
Japanese Electric Grid	Biphenyl	model/secondary	1.56e-09	1.72e-10
Japanese Electric Grid	Anthracene	model/secondary	1.55e-09	1 72e-10
Natural Gas Prod.	Chlorine	secondary	1.49e-09	1.65e-10
Fuel Oil #6 Prod	Cobalt	secondary	1 45e-09	1.60e-10
Japanese Electric Grid	Benzolg h ilnervlene	model/secondary	1.45e-09	1 60e-10
Fuel Oil #6 Prod	Benzyl chloride	secondary	1 44e-09	1.60e-10
US electric grid	Ethylbenzene	model/secondary	1 43e-09	1 59e-10
Fuel Oil #6 Prod	Aluminum (elemental)	secondary	1 37e-09	1.51e-10
Fuel Oil #2 Prod	Cadmium	secondary	1 34e-09	1 49e-10
LPG Production	Methyl tert-butyl ether	secondary	1.34e-09	1.49e-10
Monitor/module	1 1 1-Trichloroethane	nrimary	1.3 le 09	1.44e-10
Natural Gas Prod	Silicon	secondary	1.50e 09	1.44e 10
I PG Production	$Z_{inc}(+2)$	secondary	1.176-09	1.30e-10
Japanese Electric Grid	Acenanhthylene	model/secondary	1.150 09	1.23e-10
I PG Production	Phenanthrene	secondary	1.09e-09	1.23e-10
End fill #6 Prod	Chloroform	secondary	1.020.09	1.13e.10
Fuel Oil #2 Prod	Carbon digulfida	secondary	1.020-09	1.110-10
Fuel Oil #2 Flou.	Agrolain	secondary	1.00e-09	1.11e-10
L DC Braduction	Vinyl aastata	secondary	0.522.10	1.06-10
Ero rioduction	V myl acetate Propiopaldabyda	secondary	9.326-10	1.00e-10
L DC Draduction	2 Mathylahalanthrana	secondary	9.366-10	1.046-10
Let Production	Danas falsamente	secondary	9.21e-10	1.02e-10
Japanese Electric Grid	Ethylong dibrowide	model/secondary	9.1/e-10	1.02e-10
LFO Production	Lanherene	secondary	8.83e-10	9.790-11
Indural Gas Prod.	Nother all and a	secondary	8.49e-10	9.42e-11
Fuel Oil #4 Prod.	Demod shlarid	secondary	8.30e-10	9.20e-11
Fuel OII #2 Prod.	Benzyl chloride	secondary	/.6/e-10	8.51e-11

Table M-34. LCD LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Fuel Oil #2 Prod.	Cobalt	secondary	7.42e-10	8.23e-11
US electric grid	Methyl tert-butyl ether	model/secondary	7.23e-10	8.02e-11
Natural Gas Prod.	Methyl ethyl ketone	secondary	6.79e-10	7.54e-11
Japanese Electric Grid	Pyrene	model/secondary	6.73e-10	7.46e-11
Natural Gas Prod.	Tetrachloroethylene	secondary	6.71e-10	7.45e-11
Fuel Oil #4 Prod.	Di(2-ethylhexyl)phthalate	secondary	5.98e-10	6.63e-11
Natural Gas Prod.	1,2-Dichloroethane	secondary	5.97e-10	6.62e-11
Fuel Oil #2 Prod.	Aluminum (elemental)	secondary	5.96e-10	6.61e-11
Japanese Electric Grid	Fluorene	model/secondary	5.86e-10	6.50e-11
Natural Gas Prod.	Methyl methacrylate	secondary	5.81e-10	6.44e-11
LPG Production	2-Methylnaphthalene	secondary	5.76e-10	6.39e-11
Japanese Electric Grid	Fluoranthene	model/secondary	5.59e-10	6.21e-11
US electric grid	Phenol	model/secondary	5.51e-10	6.11e-11
Fuel Oil #2 Prod.	Chloroform	secondary	5.41e-10	6.00e-11
Fuel Oil #6 Prod.	Acrolein	secondary	5.33e-10	5.91e-11
US electric grid	Vinyl acetate	model/secondary	5.15e-10	5.71e-11
Natural Gas Prod.	Styrene	secondary	5.13e-10	5.69e-11
LPG Production	Chlorine	secondary	5.11e-10	5.66e-11
Natural Gas Prod.	Barium cmpds	secondary	5.06e-10	5.61e-11
Fuel Oil #2 Prod.	Propionaldehyde	secondary	5.00e-10	5.55e-11
US electric grid	Phenanthrene	model/secondary	4.85e-10	5.38e-11
Natural Gas Prod.	Bromoform	secondary	4.79e-10	5.31e-11
US electric grid	Ethylene dibromide	model/secondary	4.78e-10	5.30e-11
Fuel Oil #4 Prod.	Acetaldehyde	secondary	4.76e-10	5.28e-11
Natural Gas Prod.	Dichloromethane	secondary	4.73e-10	5.24e-11
Fuel Oil #4 Prod.	Mercury	secondary	4.46e-10	4.94e-11
Fuel Oil #6 Prod.	Methyl chloride	secondary	4.42e-10	4.91e-11
US electric grid	Xylene (mixed isomers)	model/secondary	4.27e-10	4.74e-11
Natural Gas Prod.	Chlorobenzene	secondary	3.83e-10	4.25e-11
US electric grid	Chromium (III)	model/secondary	3.82e-10	4.23e-11
LPG Production	1,1,1-Trichloroethane	secondary	3.63e-10	4.03e-11
LPG Production	Acenaphthene	secondary	3.37e-10	3.74e-11
Fuel Oil #4 Prod.	Dimethyl sulfate	secondary	3.31e-10	3.67e-11
Fuel Oil #6 Prod.	Di(2-ethylhexyl)phthalate	secondary	3.19e-10	3.53e-11
Natural Gas Prod.	2.4-Dinitrotoluene	secondary	3.07e-10	3.41e-11
Fuel Oil #4 Prod.	Toluene	secondary	2.92e-10	3.24e-11
Fuel Oil #2 Prod.	Acrolein	secondary	2.84e-10	3.15e-11
Natural Gas Prod.	Acetophenone	secondary	2.82e-10	3.13e-11
Natural Gas Prod.	3-Methylcholanthrene	secondary	2.72e-10	3.02e-11
LPG Production	Ethyl Chloride	secondary	2.57e-10	2.85e-11
Natural Gas Prod	Chromium (III)	secondary	2.54e-10	2.82e-11
Fuel Oil #6 Prod	Acetaldehyde	secondary	2.54e-10	2.82e-11
Fuel Oil #4 Prod	Cadmium cmpds	secondary	2.5 te 10 2.44e-10	2.010 11 2.71e-11
Fuel Oil #6 Prod	Mercury	secondary	2 36e-10	2.62e-11
Fuel Oil #2 Prod	Methyl chloride	secondary	2.36e-10	2.62e-11
US electric grid	2.3.7.8-TCDF	model/secondary	2.36e-10	2.50e-11
LPG Production	Cumene	secondary	2.200-10 2.18e-10	2.550 11 2.41e-11
US electric grid	1 1 1-Trichloroethane	model/secondary	2.10c-10 2.10c-10	2.330-11
LPG Production	Benzo[a]nvrene	secondary	2.100-10 2.06e-10	2.330-11 2.28e-11
Fuel Oil #6 Prod	Dimethyl sulfate	secondary	1.76e.10	1.96e-11
Natural Gas Prod	2-Methylnanhthalene	secondary	1.706-10	1.900-11
Ivatural Gas I IOu.	2-mouly maphulaiche	secondary	1./10-10	1.090-11

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Fuel Oil #2 Prod.	Di(2-ethylhexyl)phthalate	secondary	1.70e-10	1.88e-11
Japanese Electric Grid	5-Methyl chrysene	model/secondary	1.69e-10	1.88e-11
Fuel Oil #4 Prod.	Isophorone	secondary	1.60e-10	1.78e-11
Fuel Oil #4 Prod.	1,4-Dichlorobenzene	secondary	1.57e-10	1.74e-11
Natural Gas Prod.	Ethylbenzene	secondary	1.52e-10	1.69e-11
US electric grid	Acenaphthene	model/secondary	1.47e-10	1.63e-11
Fuel Oil #6 Prod.	Toluene	secondary	1.43e-10	1.58e-11
US electric grid	Ethyl Chloride	model/secondary	1.39e-10	1.54e-11
Fuel Oil #2 Prod.	Acetaldehyde	secondary	1.35e-10	1.50e-11
Natural Gas Prod.	Aluminum (elemental)	secondary	1.34e-10	1.49e-11
LPG Production	Biphenyl	secondary	1.30e-10	1.44e-11
Fuel Oil #4 Prod.	Methyl ethyl ketone	secondary	1.28e-10	1.42e-11
Fuel Oil #2 Prod.	Mercury	secondary	1.27e-10	1.41e-11
Fuel Oil #4 Prod.	Tetrachloroethylene	secondary	1.27e-10	1.40e-11
US electric grid	Cumene	model/secondary	1.18e-10	1.31e-11
Fuel Oil #4 Prod.	1,2-Dichloroethane	secondary	1.13e-10	1.25e-11
LPG Production	Lead cmpds	secondary	1.12e-10	1.24e-11
Natural Gas Prod.	Phenanthrene	secondary	1.11e-10	1.23e-11
Fuel Oil #4 Prod.	Methyl methacrylate	secondary	1.09e-10	1.21e-11
LPG Production	Dibenzo[a,h]anthracene	secondary	1.07e-10	1.18e-11
LPG Production	Acenaphthylene	secondary	1.03e-10	1.14e-11
LPG Production	Anthracene	secondary	1.01e-10	1.12e-11
Fuel Oil #4 Prod.	Aromatic hydrocarbons	secondary	9.84e-11	1.09e-11
Fuel Oil #4 Prod.	Styrene	secondary	9.67e-11	1.07e-11
LPG Production	Nickel cmpds	secondary	9.46e-11	1.05e-11
Fuel Oil #2 Prod.	Dimethyl sulfate	secondary	9.40e-11	1.04e-11
LPG Production	Benzo[a]anthracene	secondary	9.15e-11	1.01e-11
Fuel Oil #2 Prod.	Cadmium cmpds	secondary	9.10e-11	1.01e-11
Fuel Oil #4 Prod.	Bromoform	secondary	9.03e-11	1.00e-11
Fuel Oil #4 Prod.	Dichloromethane	secondary	8.91e-11	9.89e-12
Fuel Oil #2 Prod.	Toluene	secondary	8.85e-11	9.82e-12
Fuel Oil #6 Prod.	Isophorone	secondary	8.53e-11	9.46e-12
Fuel Oil #6 Prod.	Cadmium cmpds	secondary	8.00e-11	8.87e-12
Natural Gas Prod.	Methyl tert-butyl ether	secondary	7.62e-11	8.45e-12
Fuel Oil #4 Prod.	Chlorobenzene	secondary	7.22e-11	8.01e-12
LPG Production	Benzo[b,j,k]fluoranthene	secondary	7.06e-11	7.83e-12
Fuel Oil #6 Prod.	1,4-Dichlorobenzene	secondary	7.05e-11	7.82e-12
US electric grid	Biphenyl	model/secondary	7.03e-11	7.79e-12
Fuel Oil #6 Prod.	Methyl ethyl ketone	secondary	6.82e-11	7.57e-12
LPG Production	Indeno(1,2,3-cd)pyrene	secondary	6.80e-11	7.55e-12
Fuel Oil #6 Prod.	Tetrachloroethylene	secondary	6.75e-11	7.48e-12
Fuel Oil #6 Prod.	1,2-Dichloroethane	secondary	6.00e-11	6.65e-12
Fuel Oil #6 Prod.	Methyl methacrylate	secondary	5.83e-11	6.47e-12
Fuel Oil #4 Prod.	2,4-Dinitrotoluene	secondary	5.79e-11	6.42e-12
LPG Production	Chrysene	secondary	5.78e-11	6.41e-12
Natural Gas Prod.	Vinyl acetate	secondary	5.43e-11	6.02e-12
Fuel Oil #4 Prod.	Acetophenone	secondary	5.32e-11	5.90e-12
Fuel Oil #6 Prod.	Styrene	secondary	5.15e-11	5.72e-12
Natural Gas Prod.	Ethylene dibromide	secondary	5.03e-11	5.58e-12

 Table M-34.
 LCD LCIA Results for the Chronic Public Health Effects Impact Category

Description L4-Dichlorobenzere secondary 5.02-11 5.57e-12 LPG Production HALON-1301 secondary 4.81e-11 5.40e-12 LPG Production Biomoform secondary 4.81e-11 5.34e-12 Patel OH 46 Prod. Deblorome have secondary 4.45e-11 5.34e-12 Patel OH 42 Prod. Copper (-14.8-2) secondary 4.45e-11 5.34e-12 Patel OH 42 Prod. Chromium (III) secondary 4.45e-11 4.32e-12 Us electric grid Betro(bi_g/informethane model/secondary 4.14e-11 4.37e-11 Vest electric grid Betro(bi_g/informethane model/secondary 4.12e-11 4.37e-12 Vest electric grid Anthracen model/secondary 3.38e-11 4.23e-12 Vest electric grid Anthracen model/secondary 3.38e-11 4.03e-12 Vest electric grid Anthracen model/secondary 3.38e-11 4.03e-12 Vest electric grid Anthracen model/secondary 3.38e-11 4.03e-12 Vest electric grid <th>Process Group</th> <th>Material</th> <th>LCI Data Type</th> <th>Chronic Public</th> <th>% of Total</th>	Process Group	Material	LCI Data Type	Chronic Public	% of Total
Lik Ori 2 Tok. FriedMathematic Becomdary 4.878-11 5.214-12 Lik OF Production HAL (NE-1501) secondary 4.878-11 5.346-12 Lik OF Production Beronnform secondary 4.878-11 5.346-12 Lik OF Production Secondary 4.56-11 5.164-12 Lik OF Production Secondary 4.464-11 4.596-12 Lik OF Production Secondary 4.464-11 4.864-12 US electric grid Denco[apyrene model/secondary 4.146-11 4.596-12 US electric grid Berno (a) (Ji Orenntheme model/secondary 3.856-11 4.258-12 US electric grid Antonacto (phreare secondary 3.856-11 4.058-12 US electric grid Antonacto (phreare secondary 3.66-11 <	Eugl Oil #2 Prod	1.4 Dichlorobenzene	secondary	Toxicity (tox-kg)	5 57e 12
List Francisco Reconstruct Reconstruct Reconstruct Led Oil 16 Prod. Dishloromethane secondary 4.818-11 5.348-12 Pacel Oil 14 Prod. Copper (-14.8-2) secondary 4.660-11 5.758-11 5.516-12 Pacel Oil 14 Prod. Isophorone secondary 4.46-11 5.458-11 5.652-11 6.652-	I DG Production	HALON 1301	secondary	J.020-11	5.40e 12
List on Prod. Dickhormechane accondary 4.78-11 5.78-12 Fuel Oil # Prod. Copper (+1.8.+2) secondary 4.55-11 5.08-12 Fuel Oil # Prod. Sophorone secondary 4.55-11 5.08-12 US dectric gird Accemphilylene model/secondary 4.37-11 4.38-21 US electric gird Benzolplyrene model/secondary 4.37-11 4.38-21 US electric gird Benzolplyrene model/secondary 4.38-11 4.27-12 US electric gird Accemphilylene model/secondary 3.38-11 4.22-12 US electric gird Athreace model/secondary 3.38-11 4.22-12 US electric gird Athreace model/secondary 3.38-11 4.23-12 US electric gird Athreace secondary 3.66-11 4.05-12 US Olectric gird Athreace secondary 3.66-11 4.05-12 US Olectric gird Methyl electrone secondary 3.16-11 3.57-12 Ed Oli # Prod. Achreal bylera secondary<	Evel Oil #6 Prod	Bromoform	secondary	4.81e 11	5 3/e 12
Date Only Prod. Distribution Secondary 4.53e11 5.12e12 Faid Only Prod. Copper (-1 & 2-2) secondary 4.45e-11 5.10e12 Faid Only Prod. Chrominn (III) secondary 4.44e-11 4.92e-12 US electric grid Benzo[a]kj[h1oranthene model/secondary 4.12e-11 4.57e-12 US electric grid Benzo[a]kj[h1oranthene model/secondary 3.82e-11 4.27e-12 US electric grid Antinacene model/secondary 3.82e-11 4.27e-12 US electric grid Antinacene model/secondary 3.82e-11 4.37e-12 US electric grid Antinacene secondary 3.66e-11 4.05e-12 US electric grid Methyl ethyl ketone secondary 3.64e-11 4.05e-12 Pad Oll #2 Prod. Teachloroschylene secondary 3.28e-11 3.58e-12 Pad Oll #2 Prod. Methyl ethyl ketone secondary 3.28e-11 3.58e-12 Pad Oll #2 Prod. Acontachloroschane secondary 3.28e-11 3.53e-12 Pad Oll #2 Prod	Fuel Oil #6 Prod	Dichloromathana	secondary	4.810-11	5 270 12
Tat. One Prod. Explore (Trice 2) Secondary 4.55e-11 5.16-12 Fad Oil # Prod. Sophorone Secondary 4.55e-11 5.04e-12 US clearitis grid Renny [https://inter. model/secondary 4.12e-11 4.59e-12 US clearitis grid Benzolph, [k] furnamhane model/secondary 4.12e-11 4.57e-12 US clearitis grid Benzolph, [k] furnamhane model/secondary 4.12e-11 4.57e-12 US clearitis grid Antinacene model/secondary 3.68e-11 4.05e-12 Evel Oil # 2 Prod. Anonatic hydrocarbons secondary 3.66e-11 4.05e-12 Evel Oil # 2 Prod. Menly clefty l keone secondary 3.59e-11 4.39e-12 Evel Oil # 2 Prod. Menly clefty l keone secondary 3.20e-11 3.57e-12 Evel Oil # 6 Prod. Aromatic hydrocarbons secondary 3.20e-11 3.55e-12 Evel Oil # 6 Prod. Aromatic hydrocarbons secondary 3.11e-11 3.45e-12 Evel Oil # 6 Prod. Aromatic hydrocarbons secondary 3.11e-11 3.4	Fuel Oil #4 Prod	Copper (+1 & +2)	secondary	4.750-11	5.11e.12
Incl. Only 1002. Prodel. Chronium (III) secondary 4.44e-11 4.92e-12 US electric grid Accaraphthylene model/secondary 4.14e-11 4.87e-12 US electric grid Benzolpi,k/flueranthene model/secondary 4.12e-11 4.57e-12 US electric grid Benzolpi,k/flueranthene model/secondary 3.82e-11 4.22e-12 Evel Oil #0 Prod. Chlorobenzene secondary 3.82e-11 4.22e-12 Evel Oil #0 Prod. Anthracene model/secondary 3.64e-11 4.05e-12 Evel Oil #2 Prod. Tetrachloroshylene secondary 3.46e-11 4.03e-12 Evel Oil #2 Prod. Tetrachloroshylene secondary 3.30e-11 3.77e-12 Evel Oil #2 Prod. Tetrachloroshylene secondary 3.30e-11 3.57e-12 Evel Oil #2 Prod. Accomplayment secondary 3.37e-11 3.5fe-12 Evel Oil #2 Prod. Accomplayment secondary 3.11e-11 3.45e-12 Evel Oil #2 Prod. Methyl methacrylate secondary 3.11e-11 3.45e-12	Fuel Oil #2 Prod	Lisophorone	secondary	4.55e 11	5.04e.12
Lick Original Constrainty 4.742-11 4.742-11 4.742-11 4.742-11 4.742-11 4.742-11 4.742-11 4.742-11 4.742-11 4.742-11 4.742-11 4.752-11 4.752-11 4.752-11 4.752-11 4.757-11 4.757-12 US electric grid Bernzo[a]pyrene model/secondary 3.852-11 4.272-12 US electric grid Antinaceme model/secondary 3.852-11 4.272-12 LICO Production Bernzo[b]fluoranthene secondary 3.662-11 4.076-12 LPG Droduction Bernzo[b]fluoranthene secondary 3.664-11 4.036-12 LPG Droduction Merary compounds secondary 3.302-11 3.992-12 LPG Droduction Merary compounds secondary 3.302-11 3.992-12 LPG Droduction Horary compounds secondary 3.102-11 3.552-12 LPG Dridored A.compit/b alcocachons secondary 3.102-11 3.452-12 LPG Dridored Bernzo[a]pyrene secondary 3.102-11 3.452-12 LPG Dridored Bernzo[a]pyre	Fuel Oil #2 Prod	Chromium (III)	secondary	4.550-11	1.020.12
Observing pindAccelulation processDot Sectric gridBernol [bi]prenemodel/secondary4.12e-114.59e-12US electric gridBernol [bi]prenemodel/secondary3.82e-114.27e-12US electric gridAnthracenemodel/secondary3.82e-114.23e-12US electric gridAnthracenemodel/secondary3.66e-114.07e-12US electric gridAnthracenesecondary3.66e-114.07e-12Pael Oil # Prod.Bernol [bi]broarthenesecondary3.64e-114.07e-12Pael Oil # 2 Prod.Tetrachloredhylenesecondary3.04e-113.77e-12Pael Oil # 2 Prod.Mercury compoundssecondary3.20e-113.55e-12Pael Oil # 2 Prod.1.2-Dichlorrednnesecondary3.10e-113.51e-12Pael Oil # 0 Prod.1.2-Dichlorednnesecondary3.10e-113.51e-12Pael Oil # 0 Prod.Bernol [bi]yrenesecondary3.10e-113.51e-12Natural Gas Prod.Bernol [bi]yrenesecondary3.10e-113.45e-12Pael Oil # 4 Prod.2.4-Dinitrothancesecondary3.10e-113.42e-12Pael Oil # 4 Prod.Eith]benzenesecondary2.80e-113.11e-12Pael Oil # 4 Prod.Eith]benzenesecondary2.80e-113.11e-12Pael Oil # 4 Prod.Eith]benzenesecondary2.80e-113.11e-12Pael Oil # 4 Prod.Eith]benzenesecondary2.80e-113.10e-12Pael Oil # 4 Prod.Actenphetnesecondary	LUS alastria grid	A commethylong	model/secondem	4.440-11	4.926-12
Col Centri gridIntrol (a) for the condary4.16-114.15-11Vis Sekrif gridBerrol (k) (k) (horarthnenmodel (secondary)3.85-114.276-12Fuel Oil #2 Prod.Aromatic hydrocarbonssecondary3.66-114.076-12LPG ProductionBerrol (k) (k) (horarthnensecondary3.66-114.076-12LPG ProductionBerrol (k) (k) (horarthnensecondary3.66-114.076-12LPG ProductionMethyl ethyl (ktonesecondary3.66-113.996-11LPG ProductionMetryl ethyl (ktonesecondary3.206-113.576-12Fuel Oil #2 Prod.Tetrakhloroethylenesecondary3.206-113.576-12Fuel Oil #2 Prod.1.2-Dichloroethanesecondary3.206-113.516-12Fuel Oil #3 Prod.1.2-Dichloroethanesecondary3.116-113.456-12Fuel Oil #4 Prod2-Dichloroethanesecondary3.116-113.456-12Fuel Oil #4 Prodyolenesecondary3.116-113.456-12Fuel Oil #4 Prodyolenesecondary3.106-113.446-12Fuel Oil #4 Prodyolenesecondary2.806-113.106-12Fuel Oil #4 Prod.Catophenonesecondary2.806-113.106-12Fuel Oil #6 Prod.Accophenonesecondary2.806-113.106-12Vatural Gas Prod.Accophenonesecondary2.806-113.106-12Vatural Gas Prod.Othersonesecondary2.806-113.106-12Vatural Gas Prod.Accoph	US electric grid	Benzo[a]pyrene	model/secondary	4.3/6-11	4.846-12
$\begin{aligned} & \text{Dischering pin} & Distrophylophylophylophylophylophylophylophyl$	US electric grid	Denzo[h i l]fluoronthono	model/secondary	4.140-11	4.596-12
	Evel Oil #6 Pred	Chlorohongono	model/secondary	4.126-11	4.376-12
Us texting printAnnume interim index secondary $3.62-11$ $4.28+11$ LPG ProductionBenzo[b]Iluoranthenesecondary $3.66-11$ $4.05+12$ LPG ProductionMethyl ethyl katonesecondary $3.66-11$ $4.03+12$ LPG ProductionMercury compoundssecondary $3.30+11$ $3.78+12$ LPG ProductionMercury compoundssecondary $3.30+11$ $3.78+12$ LPG ProductionMercury compoundssecondary $3.20+11$ $3.58+12$ LPG ProductionFluorenesecondary $3.12+11$ $3.58+12$ LPG ProductionFluorenesecondary $3.12+11$ $3.58+12$ LPG ProductionFluorenesecondary $3.11+11$ $3.45+12$ LPG ProductionPluorenesecondary $3.11+11$ $3.45+12$ LPG ProductionPluorenesecondary $3.11+11$ $3.45+12$ LPG OH /# Prod. $o-xylene$ secondary $3.00+11$ $3.44+12$ LPu OH /# Prod. $o-xylene$ secondary $2.86+11$ $3.17+12$ LPU OH /# Prod. $A-etaphenone$ secondary $2.80+11$ $3.116+12$ Natural Gas Prod.Dihenzo[a,h]anthracenesecondary $2.80+11$ $3.116+12$ Natural Gas Prod.Dihenzo[a,h]anthracenesecondary $2.80+11$ $3.116+12$ Natural Gas Prod.Sprenesecondary $2.80+11$ $3.10+12$ Natural Gas Prod.Denzo[a,h]anthracenesecondary $2.80+11$ $3.10+12$ Natural Gas Prod.Denzo[a,h]anthracene <t< td=""><td>Fuel Oli #6 Prod.</td><td>Anthropome</td><td>secondary</td><td>3.856-11</td><td>4.27e-12</td></t<>	Fuel Oli #6 Prod.	Anthropome	secondary	3.856-11	4.27e-12
Pueb Oil #2 Prod. Aromatic hydrocarbons secondary 3.6/e-11 4.05/e-12 Fuel Oil #2 Prod. Methyl ethyl ktoine secondary 3.66e-11 4.05e-12 Fuel Oil #2 Prod. Tetrachionethylene secondary 3.59e-11 3.99e-12 ICO Production Mercury compounds secondary 3.20e-11 3.55e-12 Fuel Oil #2 Prod. 1.2-Dichloroethane secondary 3.17e-11 3.55e-12 ICO Production Fluorene secondary 3.11e-11 3.55e-12 ICO Production Fluorene secondary 3.11e-11 3.45e-12 Value Oil #2 Prod. Methyl methacrylate secondary 3.11e-11 3.45e-12 Value Oil #4 Prod. 6-xylene secondary 3.10e-11 3.45e-12 Fuel Oil #6 Prod. 2.4-Dinitrotoluene secondary 3.08e-11 3.14e-12 Fuel Oil #6 Prod. Aectophenone secondary 2.86e-11 3.11e-12 Fuel Oil #70rd. Aectophenone secondary 2.86e-11 3.11e-12 Natural Gas Prod. Aecenp		Anthracene	model/secondary	3.82e-11	4.23e-12
IPO ProductionHenzolp Informatinesecondary3.60e-114.03e-12Fuel Oil # 2 Prod.Methyl ethyl ketonesecondary3.60e-113.99e-12LPO ProductionMercury compoundssecondary3.29e-113.99e-12LPO ProductionMercury compoundssecondary3.20e-113.55e-12Fuel Oil # 6 Prod.Aromatic hydrocarbonssecondary3.12e-113.55e-12LPO ProductionFluorenesecondary3.11e-113.45e-12JWatral Gas Prod.Benzol [apyrenesecondary3.11e-113.45e-12Fuel Oil # 0 Prod.0-xylenesecondary3.11e-113.45e-12Fuel Oil # Prod.0-xylenesecondary3.10e-113.44e-12Fuel Oil # Prod.0-xylenesecondary3.10e-113.44e-12Fuel Oil # 6 Prod.2.4-Dinitrotolucnesecondary2.86e-113.17e-12Fuel Oil # 6 Prod.Acetophenonesecondary2.86e-113.17e-12Fuel Oil # 6 Prod.Acetophenonesecondary2.86e-113.11e-12Matral Gas Prod.Dibenzo[a] [anthracenesecondary2.80e-113.10e-12Matral Gas Prod.Dibenzo[a] [anthracenesecondary2.80e-113.10e-12Vol 2 electric gridBenzo[a] [anthracenesecondary2.74e-113.04e-12LPO ProductionPyrenesecondary2.74e-113.04e-12LPO ProductionBenzo[a] anthracenesecondary2.75e-112.85e-12LPO ProductionBenzo[a] anthracene </td <td>Fuel Oil #2 Prod.</td> <td>Aromatic hydrocarbons</td> <td>secondary</td> <td>3.6/e-11</td> <td>4.07e-12</td>	Fuel Oil #2 Prod.	Aromatic hydrocarbons	secondary	3.6/e-11	4.07e-12
Fuel Off #2 Prod. Methyl ethyl Rohne secondary 3.584-11 3.09-12 LPGI Off #2 Prod. Tertachileroethylene secondary 3.59-11 3.59-12 LPG Production Mercury compounds secondary 3.20-11 3.58-12 LPG Production Inz-Dichloroethane secondary 3.10-11 3.55-12 LPG Production Fluorene secondary 3.11-11 3.45-12 Natural Gas Prod. Benzo[a]pyrene secondary 3.11e-11 3.45-12 Fuel Off #2 Prod. Methyl methacrylate secondary 3.11e-11 3.45-12 Fuel Off #2 Prod. Methyl methacrylate secondary 3.11e-11 3.45-12 Fuel Off #4 Prod. 2.4-Dinitrotoluene secondary 2.80-11 3.14-12 Natural Gas Prod. Acetophenone secondary 2.80-11 3.11e-12 Natural Gas Prod. Acetophenone secondary 2.80-11 3.10e-12 Statural Gas Prod. Acetophenone secondary 2.80-11 3.10e-12 Vastural Gas Prod. Acetophenone<	LPG Production	Benzo[b]fluoranthene	secondary	3.66e-11	4.05e-12
Fuel Oil # 2 Prod. Tetrachionentylene secondary 3.398-11 3.398-11 3.398-11 3.398-12 3.388-12 3.398-12 3.388-12 3.398-12 3.388-12	Fuel Oil #2 Prod.	Methyl ethyl ketone	secondary	3.64e-11	4.03e-12
LPG Production Mercury compounds secondary 3.40e-11 3.7ke-12 Fuel Oil # Prod. Arronatic hydrocarbons secondary 3.20e-11 3.5ke-12 LPG Production Fluorene secondary 3.17e-11 3.5ke-12 LPG Production Fluorene secondary 3.11e-11 3.45e-12 Jatural Gas Prod. Berzo[a]pyrene secondary 3.11e-11 3.45e-12 Fuel Oil # 2 Prod. Methyl methacrylate secondary 3.10e-11 3.45e-12 Fuel Oil # 4 Prod. 2,4-Dinitrotoluene secondary 3.09e-11 3.42e-12 Fuel Oil # 6 Prod. 2,4-Dinitrotoluene secondary 2.88e-11 3.17e-12 Fuel Oil # 6 Prod. Actophenone secondary 2.88e-11 3.10e-12 Natural Gas Prod. Dibenzo[a,h]authracene secondary 2.80e-11 3.10e-12 Natural Gas Prod. Actophenone secondary 2.80e-11 3.01e-12 Ver Iol # 2 Prod. Styrene secondary 2.75e-11 3.05e-12 US electric grid Benzo[a]	Fuel Oil #2 Prod.	Tetrachloroethylene	secondary	3.59e-11	3.99e-12
Fuel Oil #0 Prod. Aromatic hydrocarbons secondary 3.2se-11 5.38e-12 LPG Oil #0 Prod. 1.2-Dichlororethane secondary 3.17e-11 3.51e-12 Natural Gas Prod. Benzo[a]pyrene secondary 3.11e-11 3.45e-12 Fuel Oil #2 Prod. Methyl methacrylate secondary 3.11e-11 3.45e-12 Fuel Oil #2 Prod. Oc-sylene secondary 3.10e-11 3.44e-12 Fuel Oil #4 Prod. o-sylene secondary 2.86e-11 3.14e-12 Fuel Oil #6 Prod. Cetophenone secondary 2.86e-11 3.14e-12 Natural Gas Prod. Dibenzo[a,h]anthracene secondary 2.80e-11 3.11e-12 Natural Gas Prod. Acenaphthene secondary 2.80e-11 3.10e-12 Vel Oil #2 Prod. Styrene secondary 2.75e-11 3.06e-12 US electric grid Benzo[a,h]anthracene model/secondary 2.74e-11 3.04e-12 LPG Production Prozo[a,h]anthracene secondary 2.74e-11 3.04e-12 LPG Production <	LPG Production	Mercury compounds	secondary	3.40e-11	3.77e-12
Fuel Oil #2 Prod. 1,2-Dubtroethane secondary 3,20e-11 5,35e-12 IACP Froduction Fluerenc secondary 3,11e-11 3,45e-12 Natural Gas Prod. Benzo[a]pyrene secondary 3,11e-11 3,45e-12 Fuel Oil #2 Prod. o-xylene secondary 3,10e-11 3,44e-12 Fuel Oil #6 Prod. 2,4-Dinitrotoluene secondary 2,86e-11 3,11e-12 Fuel Oil #6 Prod. Actophenone secondary 2,86e-11 3,11e-12 Natural Gas Prod. Acetophenone secondary 2,80e-11 3,11e-12 Natural Gas Prod. Acetophenone secondary 2,80e-11 3,11e-12 Natural Gas Prod. Acenaphthene secondary 2,80e-11 3,04e-12 US electric grid Benzo[a]anthracene model/secondary 2,74e-11 3,04e-12 LPG Production Pyrene secondary 2,74e-11 3,04e-12 LPG Production Benzo[a]anthracene secondary 2,74e-11 3,04e-12 LPG Production Pyrene secon	Fuel Oil #6 Prod.	Aromatic hydrocarbons	secondary	3.23e-11	3.58e-12
LPG Production Fluorene secondary 3.17e-11 3.31e-12 Fuel Oil #2 Prod. Benzo[a]pyrene secondary 3.11e-11 3.45e-12 Fuel Oil #4 Prod. o-xylene secondary 3.10e-11 3.45e-12 Fuel Oil #4 Prod. c-xylene secondary 3.09e-11 3.42e-12 Fuel Oil #6 Prod. 2.44-Dinitrotoluene secondary 2.86e-11 3.17e-12 Fuel Oil #6 Prod. Acetophenone secondary 2.83e-11 3.14e-12 Natural Gas Prod. Dibenzo[a,h]anthracene secondary 2.80e-11 3.10e-12 Vatural Gas Prod. Styrene secondary 2.75e-11 3.04e-12 US electric grid Benzo[a]anthracene model/secondary 2.74e-11 3.04e-12 LPG Production Pyrene secondary 2.74e-11 3.04e-12 LPG Production Fluoranthene secondary 2.59e-11 2.87e-12 LPG Production Fluoranthene secondary 2.57e-11 2.88e-12 LPG Production Fluoranthene secondary<	Fuel Oil #2 Prod.	1,2-Dichloroethane	secondary	3.20e-11	3.55e-12
Natural Gas Prod. Benzolajpyrene secondary 3.11e-11 3.45e-12 Fuel Oil #2 Prod. Methyl methacrylate secondary 3.10e-11 3.45e-12 Fuel Oil #4 Prod. o-xylene secondary 3.10e-11 3.42e-12 Fuel Oil #4 Prod. 2.4-Dinitrotoluene secondary 2.86e-11 3.17e-12 Fuel Oil #6 Prod. Acetophenone secondary 2.83e-11 3.11e-12 Natural Gas Prod. Dibenzo[a,h]anthracene secondary 2.80e-11 3.10e-12 Valural Gas Prod. Acetaphenone secondary 2.80e-11 3.10e-12 Valural Gas Prod. Acetaphenone secondary 2.80e-11 3.10e-12 Valural Gas Prod. Acetaphenone secondary 2.76e-11 3.04e-12 US electric grid Benzo[a]anthracene model/secondary 2.74e-11 3.04e-12 LPG Production Pyrene secondary 2.66e-11 2.95e-12 LPG Production Benzo[a,h]iperylene secondary 2.59e-11 2.85e-12 Fuel Oil #2 Prod. Bromo	LPG Production	Fluorene	secondary	3.17e-11	3.51e-12
Fuel Oil #2 Prod. Methy methacrylate secondary 3.11e-11 3.45e-12 Fuel Oil #4 Prod. 2.4-Dinitrotoluene secondary 3.09e-11 3.44e-12 Fuel Oil #6 Prod. 2.4-Dinitrotoluene secondary 2.86e-11 3.17e-12 Fuel Oil #0 Prod. A-LDinitrotoluene secondary 2.86e-11 3.17e-12 Natural Gas Prod. Acetophenone secondary 2.80e-11 3.11e-12 Natural Gas Prod. Acetophenone secondary 2.80e-11 3.10e-12 Fuel Oil #2 Prod. Styrene secondary 2.75e-11 3.05e-12 US electric grid Benzo[a]antracene mode/secondary 2.74e-11 3.04e-12 LPG Production Pyrene secondary 2.74e-11 3.04e-12 LPG Production Benzo[a],i]perylene secondary 2.57e-11 2.87e-12 LPG Production Benzo[a],i]perylene secondary 2.57e-11 2.87e-12 LPG Production Benzo[a],i]perylene secondary 2.57e-11 2.87e-12 LPG Production Benzo	Natural Gas Prod.	Benzo[a]pyrene	secondary	3.11e-11	3.45e-12
Fuel Oil #4 Prod. o-xylene secondary 3.10e-11 3.44e-12 Fuel Oil #6 Prod. 2,4-Dinitrotoluene secondary 3.09e-11 3.14e-12 Fuel Oil #4 Prod. Ethylbenzene secondary 2.88e-11 3.17e-12 Natural Gas Prod. Dibenzo[a,h]anthracene secondary 2.80e-11 3.11e-12 Natural Gas Prod. Acenaphthene secondary 2.80e-11 3.10e-12 US electric grid Benzo[a]anthracene model/secondary 2.74e-11 3.04e-12 LPG Production Pyrene secondary 2.74e-11 3.04e-12 LPG Production Benzo[a],h]perylene secondary 2.66e-11 2.95e-12 LPG Production Benzo[a],h]iperylene secondary 2.57e-11 2.87e-12 Fuel Oil #2 Prod. Bromoform secondary 2.57e-11 2.87e-12 Fuel Oil #2 Prod. Chronium (III) secondary 2.37e-11 2.87e-12 Natural Gas Prod. L,1,1-Trichloroethane secondary 2.07e-11 2.30e-12 Natural Gas Prod. C	Fuel Oil #2 Prod.	Methyl methacrylate	secondary	3.11e-11	3.45e-12
Fuel Oil #6 Prod. 2,4-Dinitrotoluene secondary 3,09e-11 3,42e-12 Fuel Oil #6 Prod. Ethylbenzene secondary 2,86e-11 3,17e-12 Natural Gas Prod. Dibenzo[a,h]anthracene secondary 2,80e-11 3,11e-12 Natural Gas Prod. Acenaphthene secondary 2,80e-11 3,11e-12 Val Oil #2 Prod. Styrene secondary 2,74e-11 3,04e-12 LPG Production Pyrene secondary 2,74e-11 3,04e-12 LPG Production Benzo[a]anthracene model/secondary 2,74e-11 3,04e-12 LPG Production Benzo[g]a.h]perylene secondary 2,57e-11 2,87e-12 LPG Production Fluoranthene secondary 2,57e-11 2,87e-12 Fuel Oil #2 Prod. Dichloromethane secondary 2,37e-11 2,87e-12 Fuel Oil #2 Prod. Dichloromethane secondary 2,37e-11 2,87e-12 Fuel Oil #2 Prod. Chromium (III) secondary 2,21e-11 2,46e-12 Natural Gas Prod. 1,1,1-Trichl	Fuel Oil #4 Prod.	o-xylene	secondary	3.10e-11	3.44e-12
Fuel Oil #4 Prod. Ethylbenzene secondary 2.86e-11 5.17e-12 Fuel Oil #6 Prod. Acetophenone secondary 2.83e-11 3.14e-12 Natural Gas Prod. Dibenzo[a,h]anthracene secondary 2.80e-11 3.11e-12 Vatural Gas Prod. Acenaphthene secondary 2.80e-11 3.10e-12 Fuel Oil #2 Prod. Styrene secondary 2.75e-11 3.05e-12 US electric grid Benzo[a,h]aptracene model/secondary 2.74e-11 3.04e-12 LPG Production Pyrene secondary 2.74e-11 3.04e-12 LPG Production Fluoranthene secondary 2.87e-11 2.87e-12 LPG Production Fluoranthene secondary 2.57e-11 2.87e-12 Fuel Oil #2 Prod. Bromoform secondary 2.37e-11 2.87e-12 Fuel Oil #2 Prod. Chornium (III) secondary 2.37e-11 2.36e-12 Natural Gas Prod. 1,1,1-Trichloroethane secondary 2.07e-11 2.30e-12 US electric grid Indeno(1,2,3-cd)pyrene<	Fuel Oil #6 Prod.	2,4-Dinitrotoluene	secondary	3.09e-11	3.42e-12
Fuel Oil #6 Prod. Acetophenone secondary 2.83e-11 3.14e-12 Natural Gas Prod. Dibenzo[a,h]anthracene secondary 2.80e-11 3.11e-12 Natural Gas Prod. Styrene secondary 2.80e-11 3.10e-12 US electric grid Benzo[a]anthracene model/secondary 2.75e-11 3.04e-12 LPG Production Pyrene secondary 2.74e-11 3.04e-12 LPG Production Benzo[g,h]]perylene secondary 2.66e-11 2.95e-11 LPG Production Bromoform secondary 2.57e-11 2.87e-12 Fuel Oil #2 Prod. Bromoform secondary 2.57e-11 2.87e-12 Fuel Oil #2 Prod. Dichloromethane secondary 2.37e-11 2.87e-12 Fuel Oil #2 Prod. Chromium (III) secondary 2.37e-11 2.63e-12 Natural Gas Prod. 1,1,1-Trichloroethane secondary 2.07e-11 2.30e-12 Vast IG as Prod. Chlorobenzene secondary 1.99e-11 2.21e-12 US electric grid Indeno(1,2,3-cd)p	Fuel Oil #4 Prod.	Ethylbenzene	secondary	2.86e-11	3.17e-12
Natural Gas Prod. Dibenzo[a,h]anthracene secondary 2.80e-11 3.11e-12 Natural Gas Prod. Acenaphthene secondary 2.80e-11 3.10e-12 Fuel Oil #2 Prod. Styrene secondary 2.75e-11 3.05e-12 US electric grid Benzo[a]anthracene model/secondary 2.74e-11 3.04e-12 LPG Production Pyrene secondary 2.74e-11 3.04e-12 LPG Production Benzo[g,h,i]perylene secondary 2.66e-11 2.95e-12 LPG Production Fluoranthene secondary 2.57e-11 2.85e-15 Fuel Oil #2 Prod. Dichloromethane secondary 2.57e-11 2.85e-12 Fuel Oil #2 Prod. Chromium (III) secondary 2.37e-11 2.63e-12 Natural Gas Prod. C, 1,1-1richloroethane secondary 2.07e-11 2.63e-12 Vatural Gas Prod. C, Chorobenzene secondary 2.07e-11 2.28e-12 Vus electric grid Indeno(1,2,3-cd)pyrene model/secondary 1.09e-12 2.16e-12 US electric grid	Fuel Oil #6 Prod.	Acetophenone	secondary	2.83e-11	3.14e-12
Natural Gas Prod. Acenaphthene secondary 2.80e-11 3.10e-12 Fuel Oil #2 Prod. Styrene secondary 2.75e-11 3.05e-12 US electric grid Benzo[a]antracene model/secondary 2.74e-11 3.04e-12 LPG Production Pyrene secondary 2.74e-11 3.04e-12 LPG Production Benzo[a],hi]perylene secondary 2.75e-11 2.87e-12 Fuel Oil #2 Prod. Bromoform secondary 2.57e-11 2.87e-12 Fuel Oil #2 Prod. Dichloromethane secondary 2.57e-11 2.87e-12 Fuel Oil #6 Prod. Chromium (III) secondary 2.37e-11 2.81e-12 Vatural Gas Prod. Chromium empds secondary 2.07e-11 2.63e-12 Natural Gas Prod. Chlorobenzene secondary 2.07e-11 2.63e-12 Vatural Gas Prod. Chlorobenzene secondary 2.07e-11 2.30e-12 US electric grid Indeno(1.2,3-cd)pyrene model/secondary 1.95e-11 2.28e-12 US electric grid Indeno(1.2,	Natural Gas Prod.	Dibenzo[a,h]anthracene	secondary	2.80e-11	3.11e-12
Fuel Oil #2 Prod.Styrenesecondary2.75e-113.05e-12US electric gridBenzo[a]anthracenemodel/secondary2.74e-113.04e-12LPG ProductionPyrenesecondary2.74e-113.04e-12LPG ProductionBenzo[g,h,i]perylenesecondary2.66e-112.95e-12LPG ProductionFluoranthenesecondary2.59e-112.87e-12Fuel Oil #2 Prod.Bromoformsecondary2.57e-112.85e-12Fuel Oil #2 Prod.Dichloromethanesecondary2.37e-112.63e-12Vatural Gas Prod.Cadmium cmpdssecondary2.21e-112.24e-12Vatural Gas Prod.Chlorobenzenesecondary2.07e-112.28e-12Fuel Oil #2 Prod.Chlorobenzenesecondary2.07e-112.28e-12Vuel Oil #2 Prod.Chlorobenzenesecondary2.05e-112.28e-12US electric gridIndeno(1,2,3-cd)pyrenemodel/secondary1.99e-112.21e-12US electric gridIndeno(1,2,3-cd)pyrenemodel/secondary1.52e-111.90e-12Fuel Oil #2 Prod.C,4-Dinitrotoluenesecondary1.52e-111.62e-12Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.52e-111.69e-12Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.52e-111.69e-12Fuel Oil #4 Prod.Copper (+1 & +2)secondary1.51e-111.68e-12Fuel Oil #4 Prod.Copper (+1 & +2)secondary1.51e-111.68e-12Fuel Oil #6 Prod.Coppe	Natural Gas Prod.	Acenaphthene	secondary	2.80e-11	3.10e-12
US electric grid Benzo[a]anthracene model/secondary 2.74e-11 3.04e-12 LPG Production Pyrene secondary 2.74e-11 3.04e-12 LPG Production Benzo[g,h,i]perylene secondary 2.66e-11 2.95e-12 LPG Production Fluoranthene secondary 2.57e-11 2.87e-12 Fuel Oil #2 Prod. Dichloromethane secondary 2.53e-11 2.87e-12 Fuel Oil #2 Prod. Dichloromethane secondary 2.37e-11 2.63e-12 Natural Gas Prod. Chromium (III) secondary 2.37e-11 2.46e-12 Natural Gas Prod. Chlorobenzene secondary 2.07e-11 2.30e-12 Fuel Oil #2 Prod. Chlorobenzene secondary 1.95e-11 2.46e-12 Natural Gas Prod. Chlorobenzene secondary 1.95e-11 2.30e-12 US electric grid Indeno(1,2,3-cd)pyrene model/secondary 1.95e-11 2.16e-12 Fuel Oil #2 Prod. Copper (+1 & +2) secondary 1.55e-11 1.82e-12 US electric grid <td< td=""><td>Fuel Oil #2 Prod.</td><td>Styrene</td><td>secondary</td><td>2.75e-11</td><td>3.05e-12</td></td<>	Fuel Oil #2 Prod.	Styrene	secondary	2.75e-11	3.05e-12
LPG ProductionPyrenesecondary2.74e-113.04e-12LPG ProductionBenzo[g,h,i]perylenesecondary2.66e-112.95e-12LPG ProductionFluoranthenesecondary2.59e-112.87e-12Fuel Oil #2 Prod.Bromoformsecondary2.57e-112.85e-12Fuel Oil #2 Prod.Dichloromethanesecondary2.57e-112.63e-12Fuel Oil #6 Prod.Chromium (III)secondary2.37e-112.63e-12Natural Gas Prod.Cadmium empdssecondary2.07e-112.30e-12Vatural Gas Prod.Chlorobenzenesecondary2.07e-112.30e-12US electric gridChrysenemodel/secondary1.99e-112.21e-12US electric gridIndeno(1,2,3-cd)pyrenemodel/secondary1.95e-112.16e-12Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.52e-111.90e-12Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.52e-111.90e-12Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.51e-111.69e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.69e-12Fuel Oil #6 Prod.Copper (+1 & +2	US electric grid	Benzo[a]anthracene	model/secondary	2.74e-11	3.04e-12
LPG ProductionBenzo[g,h,i]perylenesecondary2.66e-112.95e-12LPG ProductionFluoranthenesecondary2.59e-112.87e-12Fuel Oil #2 Prod.Bromoformsecondary2.57e-112.85e-12Fuel Oil #2 Prod.Dichloromethanesecondary2.53e-112.81e-12Fuel Oil #6 Prod.Chromium (III)secondary2.21e-112.46e-12Natural Gas Prod.Cadmium empdssecondary2.07e-112.30e-12Fuel Oil #2 Prod.Chlorobenzenesecondary2.05e-112.28e-12US electric gridChroysenemodel/secondary1.99e-112.21e-12US electric gridIndeno(1,2,3-cd)pyrenemodel/secondary1.95e-112.16e-12Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.55e-111.82e-12US electric gridFluorenemodel/secondary1.55e-111.72e-12Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.55e-111.72e-12Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.55e-111.72e-12Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.55e-111.72e-12Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.51e-111.68e-12Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.51e-111.68e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.68e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.67e-12Natural Gas Prod.Ethy	LPG Production	Pyrene	secondary	2.74e-11	3.04e-12
LPG Production Fluoranthene secondary 2.59e-11 2.87e-12 Fuel Oil #2 Prod. Bromoform secondary 2.57e-11 2.85e-12 Fuel Oil #2 Prod. Dichloromethane secondary 2.53e-11 2.81e-12 Fuel Oil #6 Prod. Chromium (III) secondary 2.37e-11 2.63e-12 Natural Gas Prod. Cadmium cmpds secondary 2.21e-11 2.46e-12 Natural Gas Prod. 1,1,1-Trichloroethane secondary 2.07e-11 2.30e-12 Fuel Oil #2 Prod. Chlorobenzene secondary 2.05e-11 2.28e-12 US electric grid Chlorobenzene model/secondary 1.99e-11 2.21e-12 US electric grid Indeno(1,2,3-cd)pyrene model/secondary 1.95e-11 2.16e-12 Fuel Oil #2 Prod. Copper (+1 & +2) secondary 1.65e-11 1.82e-12 US electric grid Indeno(1,2,3-cd)pyrene model/secondary 1.55e-11 1.72e-12 Fuel Oil #2 Prod. Copper (+1 & +2) secondary 1.65e-11 1.82e-12 Selectric gri	LPG Production	Benzo[g,h,i]perylene	secondary	2.66e-11	2.95e-12
Fuel Oil #2 Prod. Bromoform secondary 2.57e-11 2.85e-12 Fuel Oil #2 Prod. Dichloromethane secondary 2.53e-11 2.81e-12 Fuel Oil #6 Prod. Chromium (III) secondary 2.37e-11 2.63e-12 Natural Gas Prod. Cadmium empds secondary 2.07e-11 2.30e-12 Natural Gas Prod. Chlorobenzene secondary 2.07e-11 2.30e-12 VS electric grid Chrysene model/secondary 1.99e-11 2.21e-12 US electric grid Indeno(1,2,3-cd)pyrene model/secondary 1.95e-11 2.16e-12 Fuel Oil #2 Prod. Copper (+1 & +2) secondary 1.95e-11 2.16e-12 US electric grid Indeno(1,2,3-cd)pyrene model/secondary 1.95e-11 2.16e-12 Fuel Oil #2 Prod. Copper (+1 & +2) secondary 1.72e-11 1.90e-12 Fuel Oil #2 Prod. Copper (+1 & +2) secondary 1.65e-11 1.82e-12 US electric grid Fluerene model/secondary 1.51e-11 1.67e-12 Fuel Oil #6 Prod. </td <td>LPG Production</td> <td>Fluoranthene</td> <td>secondary</td> <td>2.59e-11</td> <td>2.87e-12</td>	LPG Production	Fluoranthene	secondary	2.59e-11	2.87e-12
Fuel Oil #2 Prod. Dichloromethane secondary 2.53e-11 2.81e-12 Fuel Oil #6 Prod. Chromium (III) secondary 2.37e-11 2.63e-12 Natural Gas Prod. Cadmium empds secondary 2.21e-11 2.46e-12 Natural Gas Prod. 1,1,1-Trichloroethane secondary 2.07e-11 2.30e-12 Fuel Oil #2 Prod. Chlorobenzene secondary 2.05e-11 2.28e-12 US electric grid Indeno(1,2,3-cd)pyrene model/secondary 1.99e-11 2.21e-12 US electric grid Indeno(1,2,3-cd)pyrene model/secondary 1.95e-11 2.16e-12 Fuel Oil #2 Prod. Copper (+1 & +2) secondary 1.72e-11 1.90e-12 Fuel Oil #2 Prod. Copper (+1 & +2) secondary 1.65e-11 1.82e-12 US electric grid Fluorene model/secondary 1.55e-11 1.72e-12 Fuel Oil #2 Prod. Ethylbenzene secondary 1.51e-11 1.68e-12 Fuel Oil #6 Prod. Ethylbenzene secondary 1.51e-11 1.68e-12 Fuel Oil #6	Fuel Oil #2 Prod.	Bromoform	secondary	2.57e-11	2.85e-12
Fuel Oil #6 Prod. Chromium (III) secondary 2.37e-11 2.63e-12 Natural Gas Prod. Cadmium empds secondary 2.21e-11 2.46e-12 Natural Gas Prod. 1,1,1-Trichloroethane secondary 2.07e-11 2.30e-12 Fuel Oil #2 Prod. Chlorobenzene secondary 2.05e-11 2.28e-12 US electric grid Indeno(1,2,3-cd)pyrene model/secondary 1.99e-11 2.16e-12 Fuel Oil #2 Prod. Copper (+1 & +2) secondary 1.72e-11 1.90e-12 Fuel Oil #2 Prod. Copper (+1 & +2) secondary 1.65e-11 1.82e-12 US electric grid Fluorene model/secondary 1.55e-11 1.72e-11 Fuel Oil #2 Prod. 2,4-Dinitrotoluene secondary 1.55e-11 1.72e-12 US electric grid Fluorene model/secondary 1.55e-11 1.72e-12 Fuel Oil #6 Prod. Ethylbenzene secondary 1.51e-11 1.68e-12 Fuel Oil #6 Prod. Copper (+1 & +2) secondary 1.51e-11 1.68e-12 Fuel Oil #6 Prod. </td <td>Fuel Oil #2 Prod.</td> <td>Dichloromethane</td> <td>secondary</td> <td>2.53e-11</td> <td>2.81e-12</td>	Fuel Oil #2 Prod.	Dichloromethane	secondary	2.53e-11	2.81e-12
Natural Gas Prod. Cadmium empds secondary 2.21e-11 2.46e-12 Natural Gas Prod. 1,1,1-Trichloroethane secondary 2.07e-11 2.30e-12 Fuel Oil #2 Prod. Chlorobenzene secondary 2.05e-11 2.28e-12 US electric grid Chrysene model/secondary 1.99e-11 2.21e-12 US electric grid Indeno(1,2,3-cd)pyrene model/secondary 1.95e-11 2.16e-12 Fuel Oil #2 Prod. Copper (+1 & +2) secondary 1.72e-11 1.90e-12 Fuel Oil #2 Prod. Copper (+1 & +2) secondary 1.65e-11 1.82e-12 US electric grid Fluorene model/secondary 1.55e-11 1.72e-11 US electric grid Fluorene secondary 1.55e-11 1.69e-12 US electric grid Fluorene secondary 1.51e-11 1.69e-12 Fuel Oil #6 Prod. Ethylbenzene secondary 1.51e-11 1.68e-12 Fuel Oil #2 Prod. Copper (+1 & +2) secondary 1.51e-11 1.67e-12 Natural Gas Prod. Et	Fuel Oil #6 Prod.	Chromium (III)	secondary	2.37e-11	2.63e-12
Natural Gas Prod.1,1,1-Trichloroethanesecondary2.07e-112.30e-12Fuel Oil #2 Prod.Chlorobenzenesecondary2.05e-112.28e-12US electric gridIndeno(1,2,3-cd)pyrenemodel/secondary1.99e-112.21e-12US electric gridIndeno(1,2,3-cd)pyrenemodel/secondary1.95e-112.16e-12Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.72e-111.90e-12Fuel Oil #2 Prod.2,4-Dinitrotoluenesecondary1.65e-111.82e-12US electric gridFluorenemodel/secondary1.55e-111.72e-12Fuel Oil #6 Prod.Ethylbenzenesecondary1.51e-111.69e-12Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.51e-111.68e-12Fuel Oil #6 Prod.Ethylbenzenesecondary1.51e-111.67e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.67e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.67e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.67e-12Fuel Oil #6 Prod.o-xylenesecondary1.47e-111.63e-12Fuel Oil #6 Prod.o-xylenesecondary1.46e-111.62e-12Fuel Oil #4 Prod.Methyl tert-butyl ethersecondary1.44e-111.59e-12LPG Production5-Methyl chrysenesecondary1.41e-111.57e-12	Natural Gas Prod.	Cadmium cmpds	secondary	2.21e-11	2.46e-12
Fuel Oil #2 Prod.Chlorobenzenesecondary2.05e-112.28e-12US electric gridChrysenemodel/secondary1.99e-112.21e-12US electric gridIndeno(1,2,3-cd)pyrenemodel/secondary1.95e-112.16e-12Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.72e-111.90e-12Fuel Oil #2 Prod.2,4-Dinitrotoluenesecondary1.65e-111.82e-12US electric gridFluorenemodel/secondary1.55e-111.72e-12Fuel Oil #6 Prod.Ethylbenzenesecondary1.51e-111.69e-12Fuel Oil #2 Prod.Acetophenonesecondary1.51e-111.68e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.67e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.67e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.47e-111.63e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.47e-111.63e-12Fuel Oil #6 Prod.O-xylenesecondary1.47e-111.63e-12Fuel Oil #6 Prod.o-xylenesecondary1.44e-111.52e-12Fuel Oil #4 Prod.Methyl tert-butyl ethersecondary1.44e-111.59e-12LPG Production5-Methyl chrysenesecondary1.41e-111.57e-12	Natural Gas Prod.	1,1,1-Trichloroethane	secondary	2.07e-11	2.30e-12
US electric gridChrysenemodel/secondary1.99e-112.21e-12US electric gridIndeno(1,2,3-cd)pyrenemodel/secondary1.95e-112.16e-12Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.72e-111.90e-12Fuel Oil #2 Prod.2,4-Dinitrotoluenesecondary1.65e-111.82e-12US electric gridFluorenemodel/secondary1.55e-111.72e-12Fuel Oil #6 Prod.Ethylbenzenesecondary1.51e-111.69e-12Fuel Oil #2 Prod.Acetophenonesecondary1.51e-111.68e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.67e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.47e-111.63e-12Fuel Oil #6 Prod.o-xylenesecondary1.47e-111.62e-12Fuel Oil #6 Prod.Methyl tert-butyl ethersecondary1.44e-111.59e-12Fuel Oil #4 Prod.Methyl tert-butyl ethersecondary1.44e-111.59e-12LPG Production5-Methyl chrysenesecondary1.41e-111.57e-12	Fuel Oil #2 Prod.	Chlorobenzene	secondary	2.05e-11	2.28e-12
US electric gridIndeno(1,2,3-cd)pyrenemodel/secondary1.95e-112.16e-12Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.72e-111.90e-12Fuel Oil #2 Prod.2,4-Dinitrotoluenesecondary1.65e-111.82e-12US electric gridFluorenemodel/secondary1.55e-111.72e-12Fuel Oil #6 Prod.Ethylbenzenesecondary1.52e-111.69e-12Fuel Oil #2 Prod.Acetophenonesecondary1.51e-111.68e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.67e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.67e-12Natural Gas Prod.Ethyl Chloridesecondary1.47e-111.63e-12Fuel Oil #6 Prod.o-xylenesecondary1.44e-111.62e-12Fuel Oil #4 Prod.Methyl tert-butyl ethersecondary1.44e-111.59e-12Fuel Oil #4 Prod.Secondary1.41e-111.57e-12	US electric grid	Chrysene	model/secondary	1.99e-11	2.21e-12
Fuel Oil #2 Prod.Copper (+1 & +2)secondary1.72e-111.90e-12Fuel Oil #2 Prod.2,4-Dinitrotoluenesecondary1.65e-111.82e-12US electric gridFluorenemodel/secondary1.55e-111.72e-12Fuel Oil #6 Prod.Ethylbenzenesecondary1.51e-111.69e-12Fuel Oil #2 Prod.Acetophenonesecondary1.51e-111.68e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.67e-12Natural Gas Prod.Ethyl Chloridesecondary1.47e-111.63e-12Fuel Oil #6 Prod.o-xylenesecondary1.46e-111.62e-12Fuel Oil #4 Prod.Methyl tert-butyl ethersecondary1.44e-111.59e-12LPG Production5-Methyl chrysenesecondary1.41e-111.57e-12	US electric grid	Indeno(1,2,3-cd)pyrene	model/secondary	1.95e-11	2.16e-12
Fuel Oil #2 Prod.2,4-Dinitrotoluenesecondary1.65e-111.82e-12US electric gridFluorenemodel/secondary1.55e-111.72e-12Fuel Oil #6 Prod.Ethylbenzenesecondary1.51e-111.69e-12Fuel Oil #2 Prod.Acetophenonesecondary1.51e-111.68e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.67e-12Natural Gas Prod.Ethyl Chloridesecondary1.47e-111.63e-12Fuel Oil #6 Prod.o-xylenesecondary1.47e-111.62e-12Fuel Oil #6 Prod.Methyl tert-butyl ethersecondary1.44e-111.59e-12Fuel Oil #4 Prod.Methyl tert-butyl ethersecondary1.41e-111.57e-12	Fuel Oil #2 Prod.	Copper (+1 & +2)	secondary	1.72e-11	1.90e-12
US electric gridFluorenemodel/secondary1.55e-111.72e-12Fuel Oil #6 Prod.Ethylbenzenesecondary1.52e-111.69e-12Fuel Oil #2 Prod.Acetophenonesecondary1.51e-111.68e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.67e-12Natural Gas Prod.Ethyl Chloridesecondary1.47e-111.63e-12Fuel Oil #6 Prod.o-xylenesecondary1.46e-111.62e-12Fuel Oil #4 Prod.Methyl tert-butyl ethersecondary1.44e-111.59e-12LPG Production5-Methyl chrysenesecondary1.41e-111.57e-12	Fuel Oil #2 Prod.	2,4-Dinitrotoluene	secondary	1.65e-11	1.82e-12
Fuel Oil #6 Prod.Ethylbenzenesecondary1.52e-111.69e-12Fuel Oil #2 Prod.Acetophenonesecondary1.51e-111.68e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.67e-12Natural Gas Prod.Ethyl Chloridesecondary1.47e-111.63e-12Fuel Oil #6 Prod.o-xylenesecondary1.46e-111.62e-12Fuel Oil #4 Prod.Methyl tert-butyl ethersecondary1.44e-111.59e-12LPG Production5-Methyl chrysenesecondary1.41e-111.57e-12	US electric grid	Fluorene	model/secondary	1.55e-11	1.72e-12
Fuel Oil #2 Prod.Acetophenonesecondary1.51e-111.68e-12Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.67e-12Natural Gas Prod.Ethyl Chloridesecondary1.47e-111.63e-12Fuel Oil #6 Prod.o-xylenesecondary1.46e-111.62e-12Fuel Oil #4 Prod.Methyl tert-butyl ethersecondary1.44e-111.59e-12LPG Production5-Methyl chrysenesecondary1.41e-111.57e-12	Fuel Oil #6 Prod.	Ethylbenzene	secondary	1.52e-11	1.69e-12
Fuel Oil #6 Prod.Copper (+1 & +2)secondary1.51e-111.67e-12Natural Gas Prod.Ethyl Chloridesecondary1.47e-111.63e-12Fuel Oil #6 Prod.o-xylenesecondary1.46e-111.62e-12Fuel Oil #4 Prod.Methyl tert-butyl ethersecondary1.44e-111.59e-12LPG Production5-Methyl chrysenesecondary1.41e-111.57e-12	Fuel Oil #2 Prod.	Acetophenone	secondary	1.51e-11	1.68e-12
Natural Gas Prod.Ethyl Chloridesecondary1.47e-111.63e-12Fuel Oil #6 Prod.o-xylenesecondary1.46e-111.62e-12Fuel Oil #4 Prod.Methyl tert-butyl ethersecondary1.44e-111.59e-12LPG Production5-Methyl chrysenesecondary1.41e-111.57e-12	Fuel Oil #6 Prod.	Copper (+1 & +2)	secondary	1.51e-11	1.67e-12
Fuel Oil #6 Prod.o-xylenesecondary1.46e-111.62e-12Fuel Oil #4 Prod.Methyl tert-butyl ethersecondary1.44e-111.59e-12LPG Production5-Methyl chrysenesecondary1.41e-111.57e-12	Natural Gas Prod.	Ethyl Chloride	secondary	1.47e-11	1.63e-12
Fuel Oil #4 Prod.Methyl tert-butyl ethersecondary1.44e-111.59e-12LPG Production5-Methyl chrysenesecondary1.41e-111.57e-12	Fuel Oil #6 Prod.	o-xylene	secondary	1.46e-11	1.62e-12
LPG Production5-Methyl chrysenesecondary1.41e-111.57e-12	Fuel Oil #4 Prod.	Methyl tert-butyl ether	secondary	1.44e-11	1.59e-12
	LPG Production	5-Methyl chrysene	secondary	1.41e-11	1.57e-12

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Benzo[a]anthracene	secondary	1.35e-11	1.50e-12
Natural Gas Prod.	Anthracene	secondary	1.28e-11	1.42e-12
Fuel Oil #2 Prod.	Chromium (III)	secondary	1.26e-11	1.40e-12
Natural Gas Prod.	Cumene	secondary	1.24e-11	1.38e-12
US electric grid	Fluoranthene	model/secondary	1.24e-11	1.37e-12
Natural Gas Prod.	Indeno(1,2,3-cd)pyrene	secondary	1.19e-11	1.32e-12
Fuel Oil #4 Prod.	Phenanthrene	secondary	1.13e-11	1.25e-12
LPG Production	Halogenated matter (organic)	secondary	1.12e-11	1.24e-12
Natural Gas Prod.	Acenaphthylene	secondary	1.10e-11	1.22e-12
US electric grid	Dibenzo[a,h]anthracene	model/secondary	1.06e-11	1.18e-12
US electric grid	2-Methylnaphthalene	model/secondary	1.03e-11	1.15e-12
Fuel Oil #4 Prod.	Vinyl acetate	secondary	1.02e-11	1.13e-12
US electric grid	Pyrene	model/secondary	1.02e-11	1.13e-12
Natural Gas Prod.	Benzo[b]fluoranthene	secondary	1.00e-11	1.11e-12
Fuel Oil #2 Prod.	o-xylene	secondary	9.63e-12	1.07e-12
Fuel Oil #4 Prod.	Ethylene dibromide	secondary	9.49e-12	1.05e-12
Fuel Oil #4 Prod.	Zinc (+2)	secondary	9.20e-12	1.02e-12
Natural Gas Prod.	Aromatic hydrocarbons	secondary	8.92e-12	9.90e-13
Natural Gas Prod.	Chrysene	secondary	8.56e-12	9.49e-13
Fuel Oil #4 Prod.	3-Methylcholanthrene	secondary	8.31e-12	9.22e-13
Fuel Oil #2 Prod.	Ethylbenzene	secondary	8.13e-12	9.02e-13
Fuel Oil #6 Prod.	Methyl tert-butyl ether	secondary	7.66e-12	8.49e-13
US electric grid	5-Methyl chrysene	model/secondary	7 64e-12	8 47e-13
Natural Gas Prod.	Biphenyl	secondary	7.40e-12	8.21e-13
US electric grid	o-xylene	model/secondary	7.35e-12	8.15e-13
US electric grid	Benzolg.h.ilpervlene	model/secondary	6.98e-12	7.74e-13
Fuel Oil #6 Prod	Phenanthrene	secondary	5 83e-12	6.47e-13
Fuel Oil #6 Prod	Vinyl acetate	secondary	5 45e-12	6.05e-13
Fuel Oil #4 Prod	2-Methylnaphthalene	secondary	5 20e-12	5 77e-13
Fuel Oil #6 Prod.	Ethylene dibromide	secondary	5.06e-12	5.61e-13
Natural Gas Prod	Benzolg h ilpervlene	secondary	5 03e-12	5 58e-13
Fuel Oil #4 Prod	Chlorine	secondary	4 52e-12	5 02e-13
Natural Gas Prod	Copper $(\pm 1 \& \pm 2)$	secondary	4 17e-12	4 63e-13
Fuel Oil #2 Prod	Methyl tert-butyl ether	secondary	4 08e-12	4 53e-13
Natural Gas Prod	Benzo[b i k]fluoranthene	secondary	4 03e-12	4 46e-13
Fuel Oil #4 Prod	1 1 1-Trichloroethane	secondary	3 90e-12	4 33e-13
Natural Gas Prod	Pyrene	secondary	3 86e-12	4 28e-13
Fuel Oil #6 Prod	3-Methylcholanthrene	secondary	3 73e-12	4 14e-13
Fuel Oil #6 Prod	Zinc (+2)	secondary	3.63e-12	4 02e-13
Fuel Oil #2 Prod	Phenanthrene	secondary	3 28e-12	3 64e-13
Fuel Oil #4 Prod	A cenanbthene	secondary	3.200 12 3.27e-12	3.63e-13
Fuel Oil #2 Prod	$Z_{inc}(+2)$	secondary	3.16e-12	3.51e-13
Fuel Oil #2 Prod	Vinyl acetate	secondary	2 91e-12	3.22e-13
Fuel Oil #4 Prod	Fthyl Chloride	secondary	2.510 12 2.76e-12	3.07e-13
Fuel Oil #2 Prod	Ethylene dibromide	secondary	2.700-12	2 000 12
Fuel Oil #2 Prod	3-Methylcholanthrene	secondary	2.090-12	2.990-13
Natural Gas Prod	Fluorene	secondary	2.000-12	2.750-15
Fuel Oil #4 Prod	Cumana	secondary	2.500-12	2.0/0-13
Fuel Oil #6 Prod	2 Methylnanhthalene	secondary	2.546-12	2.590-13
	2-incury maphunatene	secondary	2.546-12	2.396-13

 Table M-34.
 LCD LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public	% of Total
		Lei Duta Type	Toxicity (tox-kg)	/o or rotur
Natural Gas Prod.	Fluoranthene	secondary	2.31e-12	2.56e-13
Fuel Oil #4 Prod.	Benzo[a]pyrene	secondary	2.12e-12	2.35e-13
Fuel Oil #6 Prod.	1,1,1-Trichloroethane	secondary	2.08e-12	2.31e-13
Fuel Oil #6 Prod.	Chlorine	secondary	2.01e-12	2.23e-13
Fuel Oil #2 Prod.	2-Methylnaphthalene	secondary	1.67e-12	1.85e-13
Fuel Oil #6 Prod.	Acenaphthene	secondary	1.59e-12	1.76e-13
Fuel Oil #6 Prod.	Ethyl Chloride	secondary	1.47e-12	1.63e-13
Fuel Oil #2 Prod.	Chlorine	secondary	1.46e-12	1.62e-13
Fuel Oil #4 Prod.	Biphenyl	secondary	1.40e-12	1.55e-13
Fuel Oil #6 Prod.	Cumene	secondary	1.25e-12	1.38e-13
Fuel Oil #2 Prod.	1,1,1-Trichloroethane	secondary	1.11e-12	1.23e-13
Fuel Oil #6 Prod.	Benzo[a]pyrene	secondary	1.09e-12	1.21e-13
Fuel Oil #4 Prod.	Acenaphthylene	secondary	1.06e-12	1.18e-13
Fuel Oil #4 Prod.	Anthracene	secondary	1.02e-12	1.14e-13
Fuel Oil #2 Prod.	Acenaphthene	secondary	9.96e-13	1.10e-13
Fuel Oil #4 Prod.	Dibenzo[a,h]anthracene	secondary	9.37e-13	1.04e-13
Fuel Oil #4 Prod.	Benzo[a]anthracene	secondary	8.61e-13	9.55e-14
Fuel Oil #4 Prod.	Lead cmpds	secondary	8.20e-13	9.10e-14
Natural Gas Prod.	5-Methyl chrysene	secondary	8.05e-13	8.93e-14
Fuel Oil #2 Prod.	Ethyl Chloride	secondary	7.85e-13	8.71e-14
Fuel Oil #4 Prod.	Benzo[b,j,k]fluoranthene	secondary	7.59e-13	8.42e-14
Fuel Oil #6 Prod.	Biphenyl	secondary	7.44e-13	8.25e-14
Fuel Oil #4 Prod.	Nickel cmpds	secondary	6.93e-13	7.69e-14
Fuel Oil #2 Prod.	Cumene	secondary	6.64e-13	7.37e-14
Fuel Oil #4 Prod.	Indeno(1.2.3-cd)pyrene	secondary	6.57e-13	7.29e-14
Fuel Oil #2 Prod.	Benzo[a]pyrene	secondary	6.19e-13	6.86e-14
Fuel Oil #4 Prod.	Chrysene	secondary	5.71e-13	6.33e-14
Fuel Oil #6 Prod.	Acenaphthylene	secondary	5.49e-13	6.09e-14
Fuel Oil #6 Prod.	Anthracene	secondary	5.18e-13	5.75e-14
Fuel Oil #6 Prod.	Dibenzo[a,h]anthracene	secondary	4.07e-13	4.51e-14
Fuel Oil #6 Prod.	Benzo[a]anthracene	secondary	4.05e-13	4.50e-14
Fuel Oil #6 Prod.	Benzo[b.i.k]fluoranthene	secondary	4.04e-13	4.49e-14
Fuel Oil #2 Prod.	Biphenvl	secondary	3.96e-13	4.40e-14
Fuel Oil #4 Prod.	HALON-1301	secondary	3.57e-13	3.96e-14
Fuel Oil #4 Prod	Fluorene	secondary	3 33e-13	3 69e-14
Fuel Oil #4 Prod.	Benzo[b]fluoranthene	secondary	3.24e-13	3.59e-14
Fuel Oil #6 Prod	Indeno(1 2 3-cd)pyrene	secondary	3 18e-13	3 52e-14
Fuel Oil #2 Prod	Acenanbthylene	secondary	3 10e-13	3 43e-14
Fuel Oil #2 Prod	Dibenzo[a h]anthracene	secondary	3.06e-13	3 39e-14
Fuel Oil #2 Prod	Lead cmnds	secondary	3.06e-13	3 39e-14
Fuel Oil #2 Prod	Anthracene	secondary	3.03e-13	3.36e-14
Fuel Oil #6 Prod	Chrysene	secondary	2.82e-13	3 13e-14
Fuel Oil #4 Prod	Pyrene	secondary	2.82C-13	3.04e-14
Fuel Oil #4 Prod	Fluoranthene	secondary	2.740-13	2 000 14
Fuel Oil #6 Prod	Lead cmpds	secondary	2.0)e-13	2.59e-14
Fuel Oil #2 Prod	Benzo[a]anthracene	secondary	2.0)c-13	2.986-14
Fuel Oil #2 Prod	Nickel cmnds	secondary	2.000-13	2.570-14
Fuel Oil #4 Prod	Benzola h ilnervlene	secondary	2.300-13	2.000-14
Fuel Oil #4 Prod	Mercury compounds	secondary	2.490-13	2.776-14
Fuel Oil #6 Prod	Nickal empde	secondary	2.496-13	2.700-14
Fuel Oil #2 Prod	Panza[h i k]fluaranthana	secondary	2.276-13	2.520-14
ruei Oli #2 Prou.	Denzo[0,j,k]nuoranunene	secondary	2.10e-13	2.396-14

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Fuel Oil #2 Prod.	Indeno(1,2,3-cd)pyrene	secondary	2.01e-13	2.23e-14
Fuel Oil #6 Prod.	Fluorene	secondary	1.74e-13	1.93e-14
Fuel Oil #2 Prod.	Chrysene	secondary	1.72e-13	1.90e-14
Fuel Oil #4 Prod.	5-Methyl chrysene	secondary	1.52e-13	1.68e-14
Fuel Oil #6 Prod.	Benzo[b]fluoranthene	secondary	1.43e-13	1.58e-14
Fuel Oil #6 Prod.	Fluoranthene	secondary	1.40e-13	1.55e-14
Fuel Oil #6 Prod.	Pyrene	secondary	1.37e-13	1.52e-14
Fuel Oil #2 Prod.	HALON-1301	secondary	1.33e-13	1.47e-14
Fuel Oil #6 Prod.	HALON-1301	secondary	1.17e-13	1.30e-14
Fuel Oil #6 Prod.	Benzo[g,h,i]perylene	secondary	1.17e-13	1.29e-14
Fuel Oil #2 Prod.	Benzo[b]fluoranthene	secondary	1.05e-13	1.17e-14
Fuel Oil #2 Prod.	Fluorene	secondary	9.59e-14	1.06e-14
Fuel Oil #2 Prod.	Mercury compounds	secondary	9.27e-14	1.03e-14
Fuel Oil #4 Prod.	Halogenated matter (organic)	secondary	8.20e-14	9.10e-15
Fuel Oil #2 Prod.	Pyrene	secondary	8.17e-14	9.06e-15
Fuel Oil #6 Prod.	Mercury compounds	secondary	8.15e-14	9.04e-15
Fuel Oil #6 Prod.	5-Methyl chrysene	secondary	8.09e-14	8.97e-15
Fuel Oil #2 Prod.	Fluoranthene	secondary	7.81e-14	8.66e-15
Fuel Oil #2 Prod.	Benzo[g,h,i]perylene	secondary	7.78e-14	8.63e-15
Natural Gas Prod.	Lead cmpds	secondary	7.44e-14	8.25e-15
Natural Gas Prod.	Nickel cmpds	secondary	6.29e-14	6.97e-15
Fuel Oil #2 Prod.	5-Methyl chrysene	secondary	4.31e-14	4.78e-15
Natural Gas Prod.	HALON-1301	secondary	3.24e-14	3.59e-15
Fuel Oil #2 Prod.	Halogenated matter (organic)	secondary	3.06e-14	3.39e-15
LPG Production	Halogenated hydrocarbons (unspecified)	secondary	2.80e-14	3.11e-15
Fuel Oil #6 Prod.	Halogenated matter (organic)	secondary	2.69e-14	2.98e-15
Natural Gas Prod.	Mercury compounds	secondary	2.26e-14	2.50e-15
Natural Gas Prod.	Halogenated matter (organic)	secondary	7.44e-15	8.25e-16
Fuel Oil #4 Prod.	Halogenated hydrocarbons (unspecified)	secondary	2.05e-16	2.27e-17
Fuel Oil #2 Prod.	Halogenated hydrocarbons (unspecified)	secondary	7.64e-17	8.48e-18
Fuel Oil #6 Prod.	Halogenated hydrocarbons (unspecified)	secondary	6.72e-17	7.46e-18
Natural Gas Prod.	Halogenated hydrocarbons (unspecified)	secondary	1.86e-17	2.06e-18
Total Manufacturing			2.35e+02	2.60e+01
Use, Maintenance and Repair I	Life-cvcle Stage			
US electric grid	Sulfur dioxide	model/secondary	6.15e+02	6.82e+01
US electric grid	Nitrogen oxides	model/secondary	8.79e-01	9.75e-02
US electric grid	Methane	model/secondary	4.81e-01	5.34e-02
US electric grid	Carbon monoxide	model/secondary	4.07e-01	4.52e-02
US electric grid	Arsenic	model/secondary	2.08e-01	2.31e-02
US electric grid	Hydrochloric acid	model/secondary	1.84e-01	2.04e-02
US electric grid	PM-10	model/secondary	4.31e-02	4.78e-03
US electric grid	Selenium	model/secondary	3.46e-02	3.84e-03
US electric grid	Vanadium	model/secondary	2.61e-02	2.89e-03
US electric grid	Hydrofluoric acid	model/secondary	1.00e-02	1.11e-03
US electric grid	Formaldehvde	model/secondary	3.62e-03	4.01e-04
US electric grid	Benzene	model/secondary	2.60e-03	2.89e-04
US electric grid	Nitrous oxide	model/secondary	2.54e-03	2.81e-04
US electric grid	Phosphorus (vellow or white)	model/secondary	1 47e-03	1.63e-04
US electric grid	Zinc (elemental)	model/secondary	6 77e-04	7 51e-05
	(••••••••••)	model, beechaal y	0.770 04	

 Table M-34. LCD LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public	% of Total
			Toxicity (tox-kg)	
US electric grid	Antimony	model/secondary	5.55e-04	6.16e-05
US electric grid	Chromium (VI)	model/secondary	2.00e-04	2.22e-05
US electric grid	Molybdenum	model/secondary	1.85e-04	2.06e-05
US electric grid	Methyl hydrazine	model/secondary	1.72e-04	1.91e-05
US electric grid	2-Chloroacetophenone	model/secondary	1.61e-04	1.79e-05
US electric grid	Bromomethane	model/secondary	1.59e-04	1.77e-05
US electric grid	Nickel	model/secondary	8.64e-05	9.59e-06
US electric grid	Cyanide (-1)	model/secondary	8.37e-05	9.28e-06
US electric grid	2,3,7,8-TCDD	model/secondary	6.34e-05	7.03e-06
US electric grid	Naphthalene	model/secondary	5.11e-05	5.67e-06
US electric grid	Barium	model/secondary	4.35e-05	4.82e-06
US electric grid	Carbon disulfide	model/secondary	3.42e-05	3.80e-06
US electric grid	Benzyl chloride	model/secondary	2.62e-05	2.91e-06
US electric grid	Cadmium	model/secondary	2.04e-05	2.27e-06
US electric grid	Chloroform	model/secondary	1.85e-05	2.05e-06
US electric grid	Propionaldehyde	model/secondary	1.71e-05	1.90e-06
US electric grid	Manganese	model/secondary	1.70e-05	1.89e-06
US electric grid	Fluoride	model/secondary	1.46e-05	1.62e-06
US electric grid	Beryllium	model/secondary	1.09e-05	1.21e-06
US electric grid	Acrolein	model/secondary	9.71e-06	1.08e-06
US electric grid	Lead	model/secondary	9.52e-06	1.06e-06
US electric grid	Cobalt	model/secondary	9.07e-06	1.01e-06
US electric grid	Copper	model/secondary	8.34e-06	9.25e-07
US electric grid	Methyl chloride	model/secondary	8.05e-06	8.93e-07
US electric grid	Di(2-ethylhexyl)phthalate	model/secondary	5.80e-06	6.43e-07
US electric grid	Acetaldehyde	model/secondary	4.62e-06	5.12e-07
US electric grid	Hexane	model/secondary	4.49e-06	4.98e-07
US electric grid	Dimethyl sulfate	model/secondary	3.21e-06	3.56e-07
US electric grid	Mercury	model/secondary	2.80e-06	3.11e-07
US electric grid	Toluene	model/secondary	1.59e-06	1.76e-07
US electric grid	Isophorone	model/secondary	1.55e-06	1.72e-07
US electric grid	Methyl ethyl ketone	model/secondary	1.24e-06	1.38e-07
US electric grid	Tetrachloroethylene	model/secondary	1.23e-06	1.36e-07
US electric grid	1,2-Dichloroethane	model/secondary	1.09e-06	1.21e-07
US electric grid	Methyl methacrylate	model/secondary	1.06e-06	1.18e-07
US electric grid	Styrene	model/secondary	9.39e-07	1.04e-07
US electric grid	Bromoform	model/secondary	8.76e-07	9.72e-08
US electric grid	Dichloromethane	model/secondary	8.65e-07	9.59e-08
US electric grid	Chlorobenzene	model/secondary	7.01e-07	7.78e-08
US electric grid	2,4-Dinitrotoluene	model/secondary	5.62e-07	6.23e-08
US electric grid	Acetophenone	model/secondary	5.16e-07	5.73e-08
US electric grid	Ethylbenzene	model/secondary	2.76e-07	3.07e-08
US electric grid	Methyl tert-butyl ether	model/secondary	1.39e-07	1.55e-08
US electric grid	Phenol	model/secondary	1.06e-07	1.18e-08
US electric grid	Vinyl acetate	model/secondary	9.93e-08	1.10e-08
US electric grid	Phenanthrene	model/secondary	9.35e-08	1.04e-08
US electric grid	Ethylene dibromide	model/secondary	9.21e-08	1.02e-08
US electric grid	Xylene (mixed isomers)	model/secondary	8.23e-08	9.13e-09
US electric grid	Chromium (III)	model/secondary	7.36e-08	8.16e-09
US electric grid	2,3,7,8-TCDF	model/secondary	4.35e-08	4.83e-09
US electric grid	1,1,1-Trichloroethane	model/secondary	4.05e-08	4.49e-09
-		-		

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
US electric grid	Acenaphthene	model/secondary	2.84e-08	3.15e-09
US electric grid	Ethyl Chloride	model/secondary	2.68e-08	2.98e-09
US electric grid	Cumene	model/secondary	2.27e-08	2.52e-09
US electric grid	Biphenyl	model/secondary	1.35e-08	1.50e-09
US electric grid	Acenaphthylene	model/secondary	8.42e-09	9.34e-10
US electric grid	Benzo[a]pyrene	model/secondary	7.97e-09	8.85e-10
US electric grid	Benzo[b,j,k]fluoranthene	model/secondary	7.94e-09	8.81e-10
US electric grid	Anthracene	model/secondary	7.35e-09	8.16e-10
US electric grid	Benzo[a]anthracene	model/secondary	5.29e-09	5.86e-10
US electric grid	Chrysene	model/secondary	3.83e-09	4.25e-10
US electric grid	Indeno(1,2,3-cd)pyrene	model/secondary	3.76e-09	4.17e-10
US electric grid	Fluorene	model/secondary	2.98e-09	3.31e-10
US electric grid	Fluoranthene	model/secondary	2.38e-09	2.64e-10
US electric grid	Dibenzo[a,h]anthracene	model/secondary	2.05e-09	2.27e-10
US electric grid	2-Methylnaphthalene	model/secondary	1.99e-09	2.21e-10
US electric grid	Pyrene	model/secondary	1.97e-09	2.19e-10
US electric grid	5-Methyl chrysene	model/secondary	1.47e-09	1.63e-10
US electric grid	o-xylene	model/secondary	1.42e-09	1.57e-10
US electric grid	Benzo[g,h,i]perylene	model/secondary	1.35e-09	1.49e-10
Total Use, Maintenance and Re	pair	· ·	6.17e+02	6.84e+01
End-of-life Life-cycle Stage				
US electric grid	Sulfur dioxide	model/secondary	1.17e-01	1.29e-02
LCD landfilling	Sulfur dioxide	primary	4.45e-02	4.94e-03
LCD incineration	Sulfur dioxide	secondary	3.45e-02	3.83e-03
LCD landfilling	Carbon monoxide	primary	2.40e-03	2.66e-04
LCD landfilling	Nitrogen dioxide	primary	8.92e-04	9.90e-05
US electric grid	Nitrogen oxides	model/secondary	1.67e-04	1.85e-05
LCD landfilling	PM	primary	1.00e-04	1.11e-05
US electric grid	Methane	model/secondary	9.13e-05	1.01e-05
US electric grid	Carbon monoxide	model/secondary	7.73e-05	8.57e-06
LCD incineration	Arsenic cmpds	secondary	4.30e-05	4.77e-06
US electric grid	Arsenic	model/secondary	3.95e-05	4.38e-06
LCD landfilling	Arsenic cmpds	primary	3.65e-05	4.05e-06
US electric grid	Hydrochloric acid	model/secondary	3.49e-05	3.87e-06
LCD landfilling	Methane	primary	2.62e-05	2.91e-06
LPG Production	Carbon monoxide	secondary	2.27e-05	2.52e-06
LCD landfilling	Benzene	primary	9.74e-06	1.08e-06
LCD incineration	Lead	secondary	9.60e-06	1.06e-06
US electric grid	PM-10	model/secondary	8.18e-06	9.08e-07
LPG Production	Methane	secondary	6.61e-06	7.33e-07
US electric grid	Selenium	model/secondary	6.57e-06	7.29e-07
LPG Production	Sulfur oxides	secondary	6.31e-06	7.00e-07
US electric grid	Vanadium	model/secondary	4.94e-06	5.48e-07
LPG Production	Nitrogen oxides	secondary	4.51e-06	5.01e-07
LPG Production	Vanadium	secondary	4.12e-06	4.58e-07
LPG Production	Benzene	secondary	3.38e-06	3.75e-07
LCD landfilling	Silver compounds	primary	2.52e-06	2.79e-07
LCD incineration	Barium cmpds	secondary	2.33e-06	2.58e-07
LCD landfilling	Barium cmpds	primary	1.96e-06	2.17e-07

Table M-34. LCD LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public	% of Total
	~		Toxicity (tox-kg)	
LCD incineration	Silver compounds	secondary	1.95e-06	2.17e-07
US electric grid	Hydrofluoric acid	model/secondary	1.91e-06	2.11e-07
LCD landfilling	Hydrochloric acid	primary	1.13e-06	1.25e-07
LPG Production	PM	secondary	1.02e-06	1.13e-07
LCD landfilling	Ammonia	primary	9.17e-07	1.02e-07
LPG Production	Arsenic	secondary	8.13e-07	9.02e-08
US electric grid	Formaldehyde	model/secondary	6.87e-07	7.62e-08
LPG Production	Formaldehyde	secondary	5.28e-07	5.86e-08
US electric grid	Benzene	model/secondary	4.94e-07	5.48e-08
US electric grid	Nitrous oxide	model/secondary	4.81e-07	5.34e-08
LCD incineration	Cadmium cmpds	secondary	3.54e-07	3.92e-08
LCD landfilling	Cadmium cmpds	primary	3.27e-07	3.62e-08
US electric grid	Phosphorus (yellow or white)	model/secondary	2.78e-07	3.09e-08
LPG Production	Hydrochloric acid	secondary	1.45e-07	1.61e-08
US electric grid	Zinc (elemental)	model/secondary	1.28e-07	1.42e-08
LPG Production	Nitrous oxide	secondary	1.28e-07	1.42e-08
LCD landfilling	Hydrogen sulfide	primary	1.11e-07	1.23e-08
US electric grid	Antimony	model/secondary	1.05e-07	1.17e-08
LCD landfilling	Chromium (VI)	primary	7.27e-08	8.07e-09
LPG Production	Phosphorus (yellow or white)	secondary	3.86e-08	4.29e-09
US electric grid	Chromium (VI)	model/secondary	3.80e-08	4.22e-09
US electric grid	Molybdenum	model/secondary	3.52e-08	3.90e-09
US electric grid	Methyl hydrazine	model/secondary	3.26e-08	3.62e-09
US electric grid	2-Chloroacetophenone	model/secondary	3.06e-08	3.39e-09
US electric grid	Bromomethane	model/secondary	3.02e-08	3.35e-09
LPG Production	Fluorides (F-)	secondary	2.97e-08	3.29e-09
LPG Production	Selenium	secondary	2.92e-08	3.24e-09
LPG Production	Hydrogen sulfide	secondary	2.42e-08	2.68e-09
LPG Production	Ammonia	secondary	2.33e-08	2.58e-09
US electric grid	Nickel	model/secondary	1.64e-08	1.82e-09
US electric grid	Cvanide (-1)	model/secondary	1 59e-08	1 76e-09
US electric grid	2 3 7 8-TCDD	model/secondary	1.39e-08	1.70 c 09
LCD landfilling	Selenium	primary	1.200 00 1.05e-08	1.55 c 09
US electric grid	Nanhthalene	model/secondary	9.69e-09	1.08e-09
US electric grid	Barium	model/secondary	9.05e 09 8.25e-09	9.15e-10
I PG Production	Hydrofluoric acid	secondary	7.91e-09	8 77e-10
US electric grid	Carbon disulfide	model/secondary	6 50e-09	7.21e-10
I PG Production	Molyhdenum	secondary	6.49e.09	7.210-10
LPG Production	Chromium (VI)	secondary	6.4)e-09	6.89e-10
L PG Production	Ethane	secondary	6 11e 00	6 78e 10
LI G Hoddetion	Panzul obleride	model/secondamy	4.070.00	5 520 10
L DC Production	Zing (slamental)	model/secondary	4.978-09	5.32e-10
LPG Production		secondary	4.72e-09	5.24e-10
		. secondary	4.576-09	5.07e-10
	Carbon tetrachioride	primary	4.0/e-09	4.52e-10
	Mercury compounds	primary	3.96e-09	4.39e-10
US electric grid	Cadmium	model/secondary	3.88e-09	4.30e-10
LPG Production	Hexane	secondary	3.55e-09	3.94e-10
US electric grid	Chloroform	model/secondary	3.51e-09	3.89e-10
US electric grid	Propionaldehyde	model/secondary	3.24e-09	3.60e-10
US electric grid	Manganese	model/secondary	3.23e-09	3.58e-10
LCD landfilling	Chloroform	primary	3.18e-09	3.53e-10

LCD incinentation Chrohenterschloride secondary 2.82-09 3.16-09 LKO Production Phenol secondary 2.82-09 3.87-10 LKO Production Pentane secondary 2.26-09 2.84-10 US cleartin grid Rerythum model/secondary 2.80-10 2.20-10 US cleartin grid Acrolein model/secondary 1.84-09 2.20-10 US cleartin grid Acrolein model/secondary 1.84-09 2.20-10 US cleartin grid Cobalt model/secondary 1.84-09 2.20-10 US cleartin grid Cobalt model/secondary 1.82-09 1.92-10 US cleartin grid Cobalt model/secondary 1.82-09 1.80-10 US cleartin grid Moltyl chloride model/secondary 1.52-09 1.22-10 US cleartin grid Moltyl chloride model/secondary 8.76-10 9.72-11 US cleartin grid Moreary model/secondary 8.76-10 9.72-11 US cleartin grid Moreary model/secondary	Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
LPO Production Phenol secondary 2.78-09 3.73-2-10 LPG Production Pentane secondary 2.56-09 2.84-10 LPG Production Pentane secondary 2.66-09 2.29-10 LCD landfilling Toture primary 1.98-09 2.20-10 US electric grid Cohol model/secondary 1.84-09 2.20-10 US electric grid Load model/secondary 1.84-09 2.20-10 US electric grid Coholt model/secondary 1.72-09 1.97-10 US electric grid Coholt model/secondary 1.56-09 1.76-10 US electric grid Methyl elhoride model/secondary 1.56-09 1.76-10 US electric grid Methyl elhoride model/secondary 1.12-09 1.22-4-10 US electric grid Methyl elhoride model/secondary 8.16-10 9.72-11 US electric grid Methyl elhoride model/secondary 8.16-10 9.72-11 US electric grid Methyl aphylinthslame model/secondary	LCD incineration	Carbon tetrachloride	secondary	3.16e-09	3.50e-10
US electric grid Floride model/secondary 2.75-09 3.70-10 LPO Production Pentane secondary 2.56-09 2.24-10 LCD Inditifing Toleune primary 1.98-09 2.20-10 US electric grid Aerolein model/secondary 1.84-09 2.20-10 US electric grid Aerolein model/secondary 1.84-09 2.20-10 US electric grid Cobat model/secondary 1.72-09 1.97-10 US electric grid Cobat model/secondary 1.58-09 1.76-10 US electric grid Copper model/secondary 1.58-09 1.76-10 US electric grid Methyl elhoride model/secondary 1.72-09 1.24-10 US electric grid Actaldehyde model/secondary 8.70-10 9.72-11 US electric grid Mercany secondary 8.70-10 9.72-11 US electric grid Mercany secondary 8.70-10 9.72-11 US electric grid Mercany secondary 4.80-10 9	LPG Production	Phenol	secondary	2.82e-09	3.12e-10
LPG Production Pentane secondary 2.56e-69 2.84e-10 US cleartic grid Borylinn model/secondary 1.84e-69 2.20e-10 US cleartic grid Arovkin model/secondary 1.84e-69 2.20e-10 US eletric grid Lead model/secondary 1.84e-69 2.00e-10 LPG Production PM-10 secondary 1.77e-69 1.97e-10 US eletric grid Cobait model/secondary 1.78e-69 1.88e-69 US eletric grid Opper model/secondary 1.58e-69 1.88e-10 US eletric grid Methyl clhoride model/secondary 1.10e-09 1.22e-10 US eletric grid Di(2-ettyl/relylphthale model/secondary 8.78e-10 9.72e-11 US eletric grid Di(2-ettyl/relylphthale model/secondary 8.78e-10 9.72e-11 US eletric grid Di(2-ettyl/relylphthale model/secondary 5.78e-11 9.78e-11 US eletric grid Dimethyl sulfate model/secondary 5.78e-11 1.57e-10 5.78e-11 LCD Ina	US electric grid	Fluoride	model/secondary	2.77e-09	3.07e-10
US electric grid Beryllium model secondary 2.06-09 2.29e-10 LCD landfilling Tolucne primary 1.98-09 2.20e-10 US electric grid Lead model secondary 1.84-09 2.04e-10 US electric grid Lead model secondary 1.77e-09 1.97e-10 US electric grid Cobalt model secondary 1.72e-09 1.97e-10 US electric grid Cobpet model secondary 1.58-09 1.76e-10 US electric grid Coppet model secondary 1.58-09 1.76e-10 US electric grid Coppet model secondary 8.76e-10 9.72e-10 US electric grid Acetaldehyde model secondary 8.76e-10 9.72e-10 US electric grid Hexane model secondary 8.76e-10 9.72e-10 US electric grid Dimetryl sulfate model secondary 8.76e-10 9.72e-10 US electric grid Dimetryl sulfate model secondary 5.37e-10 5.75e-11 US electric grid Dimetryl sulfate <td< td=""><td>LPG Production</td><td>Pentane</td><td>secondary</td><td>2.56e-09</td><td>2.84e-10</td></td<>	LPG Production	Pentane	secondary	2.56e-09	2.84e-10
LCD landfilling Toluene primary 1.98-e09 2.20e-10 US electric grid Lead model/secondary 1.81e-09 2.04e-10 LPG Production PM-10 secondary 1.77e-09 1.97e-10 US electric grid Cobat model/secondary 1.72e-09 1.97e-10 LPG Production Nickel secondary 1.52e-09 1.80e-10 US electric grid Copper model/secondary 1.53e-09 1.69e-10 US electric grid Methyl chloride model/secondary 1.12e-09 1.23e-10 US electric grid Di/C-ethylhexyl pithalate model/secondary 8.76e-10 9.72e-11 US electric grid Hexane model/secondary 8.71e-10 9.73e-11 US electric grid Mercury model/secondary 8.71e-10 5.72e-11 US electric grid Mercury model/secondary 5.39e-10 5.75e-11 US electric grid Mercury model/secondary 3.39e-10 5.75e-11 US electric grid Tolenee model/secondary </td <td>US electric grid</td> <td>Beryllium</td> <td>model/secondary</td> <td>2.06e-09</td> <td>2.29e-10</td>	US electric grid	Beryllium	model/secondary	2.06e-09	2.29e-10
US electric grid Acrolein model/secondary 1.84-09 2.040-10 US electric grid Lead model/secondary 1.77e-09 1.97e-10 US electric grid Cobalt model/secondary 1.72e-09 1.97e-10 US electric grid Cobalt model/secondary 1.52e-09 1.80e-10 US electric grid Methyl chloride model/secondary 1.12e-09 1.26e-10 US electric grid Methyl chloride model/secondary 1.12e-09 1.26e-10 US electric grid Acstaldelyde model/secondary 8.16e-10 9.72e-11 US electric grid Hexane model/secondary 8.1e-10 9.72e-11 US electric grid Mercury model/secondary 6.10e-10 6.76e-11 US electric grid Mercury model/secondary 5.19e-10 5.75e-11 LCD indriftling Lead empds secondary 5.19e-10 5.75e-11 LCD indriftling Xylenc (mixed issmers) primary 4.80e-10 5.38e-11 LCD indriftling Tetrachioroethylen	LCD landfilling	Toluene	primary	1.98e-09	2.20e-10
US electric gridLeadmodel/secondary1,77e-091,97e-10LPG ProductionPM-10secondary1,77e-091,97e-10US electric gridCobaltmodel/secondary1,62e-091,80e-10US electric gridCoppermodel/secondary1,53e-091,69e-10US electric gridMethyl chloridemodel/secondary1,72e-091,80e-10US electric gridD12-e-thylhexyl phthalatemodel/secondary1,12e-091,22e-10US electric gridD12-e-thylhexyl phthalatemodel/secondary8,76e-109,73e-11US electric gridD12-e-thylhexyl phthalatemodel/secondary8,51e-109,44e-11US electric gridDimethyl sulfatemodel/secondary6,10e-106,76e-11US electric gridMercurymodel/secondary6,10e-106,76e-11LCD incinerationLead empdsprimary4,86e-105,33e-11LCD incinerationLead empdsprimary4,86e-105,33e-11LCD indfillingToluenemodel/secondary3,01e-103,34e-11US electric gridToluenemodel/secondary2,95e-103,27e-11LCD landfillingToreno/secondery2,36e-102,23e-11LCD landfillingToreno/secondery2,36e-102,32e-11US electric gridToluenemodel/secondary2,36e-102,32e-11LCD landfillingToreno/secondery2,36e-102,32e-112,36e-102,36e-10US electric gridMethyl ethyl ketonemod	US electric grid	Acrolein	model/secondary	1.84e-09	2.04e-10
LPG Production PM-10 secondary 1.72e-09 1.97e-10 US electric grid Cobult model/secondary 1.52e-09 1.80e-10 US electric grid Copper model/secondary 1.58e-09 1.76e-10 US electric grid Methyl chloride model/secondary 1.58e-09 1.76e-10 US electric grid Di2-ethylhexylphthalate model/secondary 1.10e-09 1.22e-10 US electric grid Actaidelyde model/secondary 8.51e-10 9.42e-11 US electric grid Directryl phylphthalate model/secondary 8.51e-10 9.47e-11 US electric grid Directryl solfate model/secondary 8.51e-10 9.47e-11 US electric grid Directryl solfate model/secondary 5.32e-10 5.90e-11 US electric grid Directryl solfate secondary 4.32e-10 5.33e-11 LCD indfilling Lead empds secondary 4.32e-10 5.33e-11 US electric grid Totucn model/secondary 3.01e-10 3.34e-11 US electric grid<	US electric grid	Lead	model/secondary	1.81e-09	2.00e-10
US detric gridCobaltmodel/secondary1.72-091.91e-10LPG ProductionNickelsecondary1.62e-091.80e-10US detric gridCoppermodel/secondary1.53e-091.60e-10US detric gridMethyl chloridemodel/secondary1.12e-091.23e-10LPG ProductionAturinium (-3)secondary1.12e-091.22e-10US electric gridDi(2-cthylhexyl)phthalatemodel/secondary8.76e-109.72e-11US electric gridHexanemodel/secondary8.74e-109.37e-11LPG ProductionAntimonysecondary8.51e-109.74e-11US electric gridDimethyl sulfatemodel/secondary6.10e-106.76e-11US electric gridDimethyl sulfatemodel/secondary5.32e-105.73e-11LCD indfillingLead empdssecondary4.77e-105.33e-11LCD indfillingXylenc (mixed isomers)primary4.08e-104.53e-11LCD landfillingTetrachloreethyleneprimary2.96e-103.21e-11LCD landfilling1,2-bichloreethyleneprimary2.96e-103.21e-11US electric gridTolucnemodel/secondary2.35e-103.21e-11LCD landfilling1,2-bichloreethyleneprimary2.96e-102.62e-11US electric gridTetrachloreethyleneprimary2.96e-102.62e-11US electric gridMethyl ethyl ketonemodel/secondary2.36e-102.62e-11US electric gridMethyl methace/pl	LPG Production	PM-10	secondary	1.77e-09	1.97e-10
LPG ProductionNickelsecondary1.62c-991.48e-10US electric gridMethyl chloridemodel/secondary1.53e-091.76e-10US electric gridMethyl chloridemodel/secondary1.13e-091.22e-10US electric gridDi(2-ethylhexyl)phthalatemodel/secondary8.76e-109.72e-10US electric gridAcctaldehydemodel/secondary8.76e-109.72e-11US electric gridHexanemodel/secondary8.44e-109.37e-11LPG ProductionAntimonysecondary8.44e-109.37e-11US electric gridDimethyl sulfatemodel/secondary5.19e-109.75e-11US electric gridMercurymodel/secondary5.19e-105.75e-11US electric gridMercurymodel/secondary5.19e-105.75e-11LCD landfillingLead empdssecondary4.77e-105.29e-11LCD landfillingKylene (mixed isomers)primary4.08e-103.34e-11US electric gridToluenemodel/secondary2.96e-103.27e-11LCD landfillingToluenemodel/secondary2.96e-103.27e-11US electric gridToluenemodel/secondary2.96e-103.27e-11LCD landfillingToluenemodel/secondary2.96e-103.27e-11US electric gridMethyl ethyl ketonemodel/secondary2.36e-102.26e-11US electric gridTotherhorethylenemodel/secondary2.36e-102.26e-11US electric gridTotherho	US electric grid	Cobalt	model/secondary	1.72e-09	1.91e-10
US electric gridCoppermodel/secondary1.58-091.76e-10US electric gridMethyl chloridemodel/secondary1.152-091.69e-10US electric gridDi(2-ethylhexyl)phthalatemodel/secondary8.10e-091.22e-10US electric gridHexanemodel/secondary8.76e-109.72e-11US electric gridHexanemodel/secondary8.51e-109.74e-11US electric gridHexanemodel/secondary8.51e-109.74e-11US electric gridDimethyl sulfatemodel/secondary6.10e-106.76e-11US electric gridMercurymodel/secondary5.32e-105.75e-11LCD indrillingLead empdssecondary4.80e-105.33e-11LCD indrillingKaleoriasecondary4.77e-105.32e-10LCD indrillingXylenc (mixed isomers)primary4.08e-103.34e-11US electric gridToluenemodel/secondary3.01e-103.34e-11US electric gridIsophoronemodel/secondary2.32e-103.21e-11LCD landfilling1.2-Dichoroethaneprimary2.77e-103.07e-11US electric gridMethyl ethyl ketonemodel/secondary2.33e-102.52e-11LCD landfilling1.2-Dichoroethanemodel/secondary2.33e-102.52e-11US electric gridMethyl methacrylatemodel/secondary2.36e-102.42e-11US electric gridMethyl methacrylatemodel/secondary1.36e-101.84e-11US electric gr	LPG Production	Nickel	secondary	1.62e-09	1.80e-10
US electric gridMethyl chloridemodel/secondary1.53e-091.69e-10LPG ProductionAluminum (+3)secondary1.10e-091.22e-10US electric gridDij2-erlylhexyl.phthalatemodel/secondary8.76e-109.72e-11US electric gridAcetaldehydemodel/secondary8.51e-109.44e-11US electric gridHexanemodel/secondary8.51e-109.44e-11US electric gridDimethyl sulfatemodel/secondary6.10e-106.76e-11US electric gridMercurymodel/secondary5.10e-106.76e-11US electric gridMercurymodel/secondary5.10e-105.78e-11LCD indriftlingLead empdssecondary5.10e-105.78e-11LCD landfillingXylene (nixed isomers)primary4.80e-104.33e-11US electric gridToluenemodel/secondary3.01e-103.34e-11US electric gridIsophoronemodel/secondary2.95e-103.21e-11US electric gridMethyl ethyl ketonemodel/secondary2.33e-102.59e-11US electric gridMethyl ethyl ketonemodel/secondary1.32e-10 <t< td=""><td>US electric grid</td><td>Copper</td><td>model/secondary</td><td>1.58e-09</td><td>1.76e-10</td></t<>	US electric grid	Copper	model/secondary	1.58e-09	1.76e-10
LPG ProductionAluminum (+3)secondary1.12e-091.24e-10US electric gridDi(2-ethylhexyl)phthalatemodel/secondary1.10e-091.22e-10US electric gridAcetaldehydemodel/secondary8.51e-109.47e-11US electric gridHexanemodel/secondary8.51e-109.44e-11LPG ProductionAntimonysecondary8.44e-109.37e-11US electric gridDimethyl sulfatemodel/secondary6.10e-106.76e-11US electric gridMercurymodel/secondary5.32e-105.90e-11LCD indfillingLead empdsprimary4.80e-105.33e-11LCD landfillingKale anpdsprimary4.80e-105.32e-11LCD landfillingXylene (mixed isomers)primary4.08e-105.32e-11US electric gridTolenemodel/secondary3.01e-103.34e-11US electric gridIsophoronemodel/secondary2.95e-103.27e-11US electric gridIsophoroneprimary2.77e-103.07e-11US electric gridMethyl ethyl ketonemodel/secondary2.33e-102.62e-11US electric gridMethyl methacrylatemodel/secondary2.30e-102.30e-11US electric gridMethyl methacrylatemodel/secondary1.72e-101.91e-11US electric gridMethyl methacrylatemodel/secondary1.72e-101.91e-11US electric gridBromoformmodel/secondary1.72e-101.91e-11US electric grid <td< td=""><td>US electric grid</td><td>Methyl chloride</td><td>model/secondary</td><td>1.53e-09</td><td>1.69e-10</td></td<>	US electric grid	Methyl chloride	model/secondary	1.53e-09	1.69e-10
US electric gridDi(2-ethylhexyl)phthalatemodel/secondary1.10e-091.22e-10US electric gridAcetaldehydemodel/secondary8.76e-109.72e-11US electric gridHexnemodel/secondary8.51e-109.44e-11US electric gridDimethyl sulfatemodel/secondary6.10e-106.76e-11US electric gridMercurymodel/secondary5.32e-105.99e-11LCD landfillingLead empdssecondary5.19e-105.75e-11LCD landfillingXplane (mixed isomers)primary4.80e-104.35a-11US electric gridToluenemodel/secondary3.01e-103.34e-11US electric gridIolenemodel/secondary3.01e-103.34e-11US electric gridIsophoronemodel/secondary2.95e-103.27e-11US electric gridIsophoronemodel/secondary2.33e-102.52e-11US electric gridMethyl ethyl ethoprimary2.95e-103.27e-11US electric gridTetrachloroethyleneprimary2.77e-103.07e-11US electric gridMethyl ethonemodel/secondary2.36e-102.25e-11US electric gridMethyl ethonemodel/secondary2.36e-102.25e-11US electric gridMethyl methacrylatemodel/secondary2.36e-102.30e-11US electric gridSyrenemodel/secondary1.72e-101.91e-11US electric gridSyrenemodel/secondary1.35e-101.52e-11LCD landfilling<	LPG Production	Aluminum (+3)	secondary	1.12e-09	1.24e-10
US electric gridAcetaldehydemodel/secondary8.76e-109.72e-11US electric gridHexanemodel/secondary8.51e-109.44e-10LPG ProductionAntimonysecondary8.44e-109.37e-11US electric gridDimethyl sulfatemodel/secondary6.10e-106.76e-11US electric gridMercurymodel/secondary5.32e-105.90e-11LCD indifilingLead empdsprimary4.80e-105.33e-11LCD landfillingLead empdsprimary4.80e-105.33e-11LCD landfillingXylene (mixed isomers)primary4.08e-103.34e-11US electric gridToluenemodel/secondary2.95e-103.27e-11LCD landfilling1.2-Dichloroethaneprimary2.90e-103.21e-11LCD landfilling1.2-Dichloroethaneprimary2.30e-103.26e-10US electric gridMethyl ethyl ketonemodel/secondary2.36e-102.26e-11US electric gridMethyl ethyl methacrylatemodel/secondary2.02e-102.24e-11US electric gridMethyl ethyl methacrylatemodel/secondary2.02e-102.24e-11US electric gridMethyl ethyl methacrylatemodel/secondary1.36e-101.88e-11US electric gridMethyl ethyl methacrylatemodel/secondary1.72e-101.91e-11US electric gridDichloromethanemodel/secondary1.35e-101.88e-11US electric gridDichloromethanemodel/secondary1.35e-101.88e-11<	US electric grid	Di(2-ethylhexyl)phthalate	model/secondary	1.10e-09	1.22e-10
US electric gridHexanemodel/secondary8.51e-109.44e-11LPO ProductionAntimonysecondary8.44e-109.37e-11US electric gridDimethyl sulfatemodel/secondary5.32e-105.76e-11US electric gridLead empdssecondary5.32e-105.75e-11LCD incinerationLead empdsprimary4.80e-105.33e-11LCD landfillingLead empdsprimary4.80e-105.33e-11LCD landfillingXylene (mixed isomers)primary4.08e-104.35e-11US electric gridToluenemodel/secondary2.90e-103.21e-11LCD landfillingTetrachloroethyleneprimary2.90e-103.21e-11LCD landfilling1,2-Dichloroethaneprimary2.90e-103.21e-11LCD landfilling1,2-Dichloroethaneprimary2.36e-102.59e-11US electric gridMethyl ethyl ktopemodel/secondary2.38e-102.59e-11US electric gridMethyl ethyl ktopemodel/secondary2.38e-102.59e-11US electric gridMethyl ethyl ktopemodel/secondary2.38e-102.30e-11US electric gridMethyl methacrylatemodel/secondary2.38e-102.30e-11US electric gridMethyl methacrylatemodel/secondary1.36e-101.88e-11LCD landfillingTrichloroethyleneprimary1.72e-101.91e-11US electric gridDichloromethaneprimary1.32e-101.92e-11LCD landfillingTric	US electric grid	Acetaldehyde	model/secondary	8.76e-10	9.72e-11
LPG ProductionAntimonysecondary8.44e-109.37e-11US electric gridDimethyl sulfatemodel/secondary6.10e-106.76e-11US electric gridMercurymodel/secondary5.32e-105.90e-11LCD landfillingLead empdssecondary5.32e-105.75r-11LCD landfillingLead empdssecondary4.80e-105.33e-11LPG ProductionNitratesecondary4.77e-105.29e-11LCD landfillingKylene (mixed isomers)primary4.08e-104.53e-11US electric gridToluenemodel/secondary3.01e-103.34e-11US electric gridIsophoronemodel/secondary2.95e-103.27e-11LCD landfilling1.2-Dichloroethaneprimary2.90e-103.21e-11US electric gridMethyl ethyl ketonemodel/secondary2.36e-102.62e-11US electric gridMethyl ethyl ketonemodel/secondary2.36e-102.62e-11US electric grid1.2-Dichloroethanemodel/secondary2.02e-102.30e-11US electric gridMethyl methacrylatemodel/secondary2.02e-102.24e-11US electric gridBryrenemodel/secondary1.78e-101.98e-11LCD landfillingTrichloroethyleneprimary1.72e-101.91e-11US electric gridBromoformmodel/secondary1.66e-101.82e-11LCD landfillingTrichloroethylenesecondary1.33e-101.48e-11US electric gridDichloromet	US electric grid	Hexane	model/secondary	8.51e-10	9.44e-11
US electric gridDimethyl sulfatemodel/secondary6.10e-106.76e-11US electric gridMercurymodel/secondary5.32e-105.90e-11LCD inndifilingLead empdssecondary5.19e-105.75e-11LCD landfillingLead empdsprimary4.80e-105.33e-11LCD forductionNitratesecondary4.77e-105.23e-11LCD landfillingXylene (mixed isomers)primary4.08e-104.53e-11US electric gridToluenemodel/secondary3.01e-103.34e-11LCD landfillingTetrachloroethyleneprimary2.90e-103.21e-11LCD landfilling1.2-Dichloroethaneprimary2.90e-103.21e-11LCD landfilling1.2-Dichloroethaneprimary2.36e-102.62e-11US electric gridMethyl ethyl tetonemodel/secondary2.36e-102.62e-11US electric gridMethyl nethyl netherylatemodel/secondary2.02e-102.30e-11US electric gridStyrenemodel/secondary2.02e-102.30e-11US electric gridBromoformmodel/secondary1.78e-101.98e-11LCD landfillingTrichloroethyleneprimary1.72e-101.91e-11US electric gridBromoformmodel/secondary1.66e-101.84e-11LCD landfillingTrichloroethylenesecondary1.35e-101.52e-11LCD forductionCoppersecondary1.35e-101.52e-11LPG ProductionSilconsecondary<	LPG Production	Antimony	secondary	8.44e-10	9.37e-11
US electric gridMercurymodel/secondary5.32e-105.90e-11LCD incincrationLead empdssecondary5.19e-105.75e-11LCD landfillingLead empdsprimary4.80e-105.33e-11LCD landfillingLead empdsprimary4.80e-105.33e-11LCD landfillingXylene (mixed isomers)primary4.08e-104.53e-11US electric gridIsophoronemodel/secondary3.01e-103.34e-11US electric gridIsophoronemodel/secondary2.95e-103.27e-11LCD landfillingTetrachloroethyleneprimary2.96e-103.27e-11LCD landfilling1,2-Dichloroethaneprimary2.36e-102.62e-11US electric gridMethyl ethyl ketonemodel/secondary2.33e-102.59e-11US electric grid1,2-Dichloroethanemodel/secondary2.07e-102.30e-11US electric grid1,2-Dichloroethanemodel/secondary2.02e-102.24e-11US electric gridStyrenemodel/secondary2.02e-102.24e-11US electric gridBromoformmodel/secondary1.78e-101.98e-11US electric gridDichloromethanemodel/secondary1.36e-101.84e-11US electric gridDichloromethanemodel/secondary1.36e-101.84e-11US electric gridDichloromethanesecondary1.36e-101.84e-11LCD landfillingTrichloroethylenesecondary1.36e-101.84e-11LPG ProductionStil	US electric grid	Dimethyl sulfate	model/secondary	6.10e-10	6.76e-11
LCD incinerationLead empdssecondary5.19e-105.75e-11LCD landfillingLead empdsprimary4.80e-105.33e-11LPG ProductionNitratesecondary4.77e-105.29e-11LCD landfillingXylene (mixed isomers)primary4.08e-103.34e-11US electric gridToluenemodel/secondary3.01e-103.34e-11US electric gridIsophoronemodel/secondary2.95e-103.27e-11LCD landfillingTetrachloroethyleneprimary2.90e-103.21e-11LCD landfilling1.2-Dichloroethanemodel/secondary2.33e-102.26e-10US electric gridMethyl ethyl ketonemodel/secondary2.33e-102.59e-11US electric gridTetrachloroethylenemodel/secondary2.07e-102.30e-11US electric gridMethyl methacrylatemodel/secondary2.02e-102.24e-11US electric gridStyrenemodel/secondary1.78e-101.98e-11LCD landfillingTrichloroethyleneprimary1.72e-101.91e-11US electric gridBromoformmodel/secondary1.66e-101.84e-11LCD landfillingTrichloroethylenesecondary1.35e-101.50e-11LPG ProductionCoppersecondary1.32e-101.52e-11LPG ProductionSiliconsecondary1.32e-101.48e-11LCD landfillingTrichloroethylenesecondary1.32e-101.48e-11LPG ProductionSiliconsecondary	US electric grid	Mercury	model/secondary	5.32e-10	5.90e-11
LCD landfillingLead cmpdsprimary4.80e-105.33e-11LPG ProductionNitratesecondary4.77e-105.29e-11LCD landfillingXylene (mixed isomers)primary4.08e-104.53e-11US electric gridToluenemodel/secondary3.01e-103.34e-11US electric gridIsophoronemodel/secondary2.95e-103.27e-11LCD landfilling1,2-Dichloroethyleneprimary2.90e-103.21e-11LCD landfilling1,2-Dichloroethyleneprimary2.36e-102.62e-11US electric gridMethyl ethyl ketonemodel/secondary2.33e-102.59e-11US electric gridTetrachloroethylenemodel/secondary2.07e-102.30e-11US electric gridMethyl methacrylatemodel/secondary2.02e-102.24e-11US electric gridStyrenemodel/secondary1.72e-101.91e-11US electric gridDichloroethanemodel/secondary1.66e-101.84e-11LCD landfillingTrichloroethyleneprimary1.72e-101.91e-11US electric gridDichloromethanemodel/secondary1.66e-101.84e-11LCD landfillingTrichloroethylenesecondary1.35e-101.52e-11LPG ProductionCoppersecondary1.32e-101.84e-11LPG ProductionSiliconsecondary1.32e-101.48e-11LCD incinerationTrichloroethylenesecondary1.32e-101.48e-11LPG ProductionSiliconse	LCD incineration	Lead cmpds	secondary	5.19e-10	5.75e-11
LPG ProductionNitratesecondary4.77c-105.29c-11LCD landfillingXylene (mixed isomers)primary4.08e-104.53e-11US electric gridToluenemodel/secondary3.01c-103.34c-11US electric gridIsophoronemodel/secondary2.95e-103.27e-11LCD landfillingTetrachloroethyleneprimary2.90e-103.21e-11LCD landfilling1,2-Dichloroethaneprimary2.36e-102.62e-11US electric gridMethyl ethyl ketonemodel/secondary2.33e-102.62e-11US electric grid1,2-Dichloroethanemodel/secondary2.33e-102.59e-11US electric grid1,2-Dichloroethanemodel/secondary2.07e-102.30e-11US electric grid1,2-Dichloroethanemodel/secondary2.02e-102.24e-11US electric gridMethyl methacrylatemodel/secondary1.07e-101.98e-11LCD landfillingTrichloroethyleneprimary1.72e-101.91e-11US electric gridBromoformmodel/secondary1.64e-101.82e-11LCD landfillingCoppersecondary1.33e-101.82e-11LPG ProductionCoppersecondary1.33e-101.48e-11LPG ProductionSiliconsecondary1.33e-101.48e-11LPG ProductionSiliconsecondary1.33e-101.48e-11LPG ProductionDimethanesecondary1.32e-101.48e-11LPG ProductionDimethylbenzanthracenesecon	LCD landfilling	Lead cmpds	primary	4.80e-10	5.33e-11
LCD landfillingXylene (mixed isomers)primary4.08e-104.53e-11US electric gridToluenemodel/secondary3.01e-103.34e-11US electric gridIsophoronemodel/secondary2.95e-103.27e-11LCD landfillingTetrachloroethyleneprimary2.90e-103.21e-11LCD landfilling1,2-Dichloroethaneprimary2.77e-103.07e-11US electric gridMethyl ethyl ketonemodel/secondary2.33e-102.59e-11US electric grid1,2-Dichloroethanemodel/secondary2.07e-102.30e-11US electric grid1,2-Dichloroethanemodel/secondary2.07e-102.30e-11US electric gridStyrenemodel/secondary2.02e-102.24e-11US electric gridStyrenemodel/secondary1.78e-101.98e-11LCD landfillingTrichloroethyleneprimary1.72e-101.91e-11US electric gridBromoformmodel/secondary1.66e-101.84e-11US electric gridDichloromethanemodel/secondary1.35e-101.50e-11LPG ProductionCoppersecondary1.35e-101.50e-11LPG ProductionSiliconsecondary1.33e-101.48e-11LCD landfillingChlorobenzenemodel/secondary1.32e-101.48e-11LPG ProductionSiliconsecondary1.32e-101.48e-11LPG ProductionDimethylenzanthracenesecondary1.25e-101.39e-11LPG ProductionDimethylenzanthrac	LPG Production	Nitrate	secondary	4.77e-10	5.29e-11
US electric gridToluenemodel/secondary3.01e-103.34e-11US electric gridIsophoronemodel/secondary2.95e-103.27e-11LCD landfilling1,2-Dichloroethyleneprimary2.90e-103.21e-11LCD landfilling1,2-Dichloroethaneprimary2.77e-103.07e-11US electric gridMethyl ethyl ketonemodel/secondary2.36e-102.62e-11US electric grid1,2-Dichloroethanemodel/secondary2.33e-102.59e-11US electric grid1,2-Dichloroethanemodel/secondary2.07e-102.30e-11US electric grid1,2-Dichloroethanemodel/secondary2.02e-102.24e-11US electric gridStyrenemodel/secondary2.02e-102.24e-11US electric gridBromofornmodel/secondary1.78e-101.98e-11LCD landfillingTrichloroethyleneprimary1.72e-101.91e-11US electric gridBromofornmodel/secondary1.66e-101.84e-11US electric gridDichloromethanemodel/secondary1.35e-101.52e-11LPG ProductionCoppersecondary1.33e-101.48e-11LPG ProductionSiliconsecondary1.33e-101.48e-11LPG ProductionChlorobenzenemodel/secondary1.25e-101.39e-11LPG ProductionDimethylenzanthracenesecondary1.25e-101.39e-11LPG ProductionDimethylenzanthracenesecondary1.25e-101.48e-11US electric grid <td>LCD landfilling</td> <td>Xylene (mixed isomers)</td> <td>primary</td> <td>4.08e-10</td> <td>4.53e-11</td>	LCD landfilling	Xylene (mixed isomers)	primary	4.08e-10	4.53e-11
US electric gridIsophoronemodel/secondary2.95e-103.27e-11LCD landfillingTetrachloroethyleneprimary2.90e-103.21e-11LCD landfilling1,2-Dichloroethaneprimary2.77e-103.07e-11US electric gridMethyl ethyl ketonemodel/secondary2.33e-102.62e-11US electric grid1,2-Dichloroethanemodel/secondary2.33e-102.62e-11US electric grid1,2-Dichloroethanemodel/secondary2.07e-102.30e-11US electric gridMethyl methacrylatemodel/secondary2.02e-102.24e-11US electric gridStyrenemodel/secondary1.78e-101.98e-11LCD landfillingTrichloroethyleneprimary1.72e-101.91e-11US electric gridBromoformmodel/secondary1.66e-101.84e-11US electric gridDichloromethanemodel/secondary1.66e-101.84e-11LPG ProductionCoppersecondary1.37e-101.52e-11LPG ProductionMethyl hydrazinesecondary1.33e-101.48e-11LCD incinerationTrichloroethylenesecondary1.33e-101.48e-11LCD incinerationTrichloroethylenesecondary1.32e-101.52e-11LPG ProductionSiliconsecondary1.33e-101.48e-11LCD incinerationTrichloroethylenesecondary1.32e-101.48e-11LPG ProductionDimethylbenzanthracenesecondary1.32e-101.39e-11LPG Production <td>US electric grid</td> <td>Toluene</td> <td>model/secondary</td> <td>3.01e-10</td> <td>3.34e-11</td>	US electric grid	Toluene	model/secondary	3.01e-10	3.34e-11
LCD landfillingTetrachloroethyleneprimary2.90e-103.21e-11LCD landfilling1,2-Dichloroethaneprimary2.77e-103.07e-11US electric gridMethyl ethyl ketonemodel/secondary2.36e-102.62e-11US electric gridTetrachloroethylenemodel/secondary2.33e-102.59e-11US electric grid1,2-Dichloroethanemodel/secondary2.07e-102.30e-11US electric gridMethyl methacrylatemodel/secondary2.07e-102.30e-11US electric gridStyrenemodel/secondary1.78e-101.98e-11LCD landfillingTrichloroethyleneprimary1.72e-101.91e-11US electric gridBromoformmodel/secondary1.66e-101.84e-11LPG ProductionCoppersecondary1.35e-101.50e-11LPG ProductionMethyl hydrazinesecondary1.33e-101.48e-11LPG ProductionSiliconsecondary1.33e-101.48e-11LCD incinerationTrichloroethylenemodel/secondary1.33e-101.48e-11LPG ProductionSiliconsecondary1.33e-101.48e-11LPG ProductionChlorobenzenemodel/secondary1.32e-101.39e-11LPG Production2-Chloroaetophenonesecondary1.32e-101.39e-11LPG ProductionBromomethanesecondary1.25e-101.39e-11LPG ProductionBromomethanesecondary1.26e-101.39e-11LPG ProductionBromomethane <td< td=""><td>US electric grid</td><td>Isophorone</td><td>model/secondary</td><td>2.95e-10</td><td>3.27e-11</td></td<>	US electric grid	Isophorone	model/secondary	2.95e-10	3.27e-11
LCD ladfilling1,2-Dichloroethaneprimary2.77e-103.07e-11US electric gridMethyl ethyl ketonemodel/secondary2.36e-102.62e-11US electric gridTetrachloroethylenemodel/secondary2.33e-102.59e-11US electric grid1,2-Dichloroethanemodel/secondary2.07e-102.30e-11US electric gridMethyl methaarylatemodel/secondary2.02e-102.24e-11US electric gridStyrenemodel/secondary1.78e-101.98e-11LCD landfillingTrichloroethyleneprimary1.72e-101.91e-11US electric gridBromoformmodel/secondary1.66e-101.84e-11US electric gridDichloromethanemodel/secondary1.37e-101.52e-11LPG ProductionCoppersecondary1.35e-101.50e-11LPG ProductionSiliconsecondary1.33e-101.48e-11US electric gridChlorobenzenemodel/secondary1.33e-101.48e-11LPG ProductionSiliconsecondary1.32e-101.48e-11LPG ProductionDimethylenesecondary1.32e-101.48e-11LPG ProductionDimethylenzanthracenesecondary1.27e-101.48e-11LPG ProductionDimethylenzanthracenesecondary1.25e-101.39e-11LPG ProductionDimethylenzanthracenesecondary1.26e-101.39e-11LPG ProductionBromomethanesecondary1.26e-101.39e-11LPG ProductionBromomethane	LCD landfilling	Tetrachloroethylene	primary	2.90e-10	3.21e-11
US electric gridMethyl ethyl ketonemodel/secondary2.36e-102.62e-11US electric gridTetrachloroethylenemodel/secondary2.33e-102.59e-11US electric grid1,2-Dichloroethanemodel/secondary2.07e-102.30e-11US electric gridMethyl methacrylatemodel/secondary2.02e-102.24e-11US electric gridStyrenemodel/secondary1.78e-101.98e-11LCD landfillingTrichloroethyleneprimary1.72e-101.91e-11US electric gridBromoformmodel/secondary1.66e-101.84e-11US electric gridDichloromethanemodel/secondary1.37e-101.52e-11LPG ProductionCoppersecondary1.35e-101.52e-11LPG ProductionSiliconsecondary1.33e-101.48e-11US electric gridChlorobenzenemodel/secondary1.33e-101.48e-11LPG ProductionSiliconsecondary1.33e-101.48e-11LPG ProductionDimethylenesecondary1.32e-101.48e-11LPG ProductionDimethylenzenesecondary1.32e-101.48e-11LPG ProductionDimethylenzenesecondary1.32e-101.48e-11LPG ProductionDimethylenzenesecondary1.32e-101.48e-11LPG ProductionDimethylenzenesecondary1.32e-101.48e-11LPG ProductionDimethylenzenesecondary1.32e-101.48e-11LPG ProductionBromomethanesecondary <td>LCD landfilling</td> <td>1.2-Dichloroethane</td> <td>primary</td> <td>2.77e-10</td> <td>3.07e-11</td>	LCD landfilling	1.2-Dichloroethane	primary	2.77e-10	3.07e-11
US electric grid Tetrachloroethylene model/secondary 2.33e-10 2.59e-11 US electric grid 1,2-Dichloroethane model/secondary 2.07e-10 2.30e-11 US electric grid Methyl methacrylate model/secondary 2.02e-10 2.24e-11 US electric grid Styrene model/secondary 1.78e-10 1.98e-11 LCD landfilling Trichloroethylene primary 1.72e-10 1.91e-11 US electric grid Bromoform model/secondary 1.66e-10 1.84e-11 US electric grid Dichloromethane model/secondary 1.66e-10 1.84e-11 LPG Production Copper secondary 1.37e-10 1.52e-11 LPG Production Methyl hydrazine secondary 1.35e-10 1.50e-11 LCD incineration Trichloroethylene secondary 1.33e-10 1.48e-11 US electric grid Chlorobenzene model/secondary 1.33e-10 1.48e-11 LPG Production Trichloroethylene secondary 1.33e-10 1.48e-11 LPG Production Dimethylene secondary 1.33e-10 1.48e-11 LPG Production Dimethylene secondary 1.33e-10 1.48e-11 LPG Production Dimethylene secondary 1.33e-10 1.48e-11 LPG Production 2-Chloroaectophenone secondary 1.27e-10 1.41e-11 LPG Production Dimethylene secondary 1.27e-10 1.41e-11 LPG Production Dimethylene secondary 1.27e-10 1.41e-11 LPG Production Methyl hydrazene secondary 1.26e-10 1.39e-11 LPG Production Dimethylene secondary 1.27e-10 1.41e-11 US electric grid 2.4-Dinitrotoluene model/secondary 1.07e-10 1.19e-11 US electric grid 2.4-Dinitrotoluene model/secondary 1.07e-10 1.19e-11 US electric grid 2.4-Dinitrotoluene model/secondary 1.07e-10 1.19e-11 US electric grid 2.4-Dinitrotoluene model/secondary 1.07e-10 1.14e-11 LPG Production Naphthalene secondary 0.79e-11 1.09e-11 US electric grid 2.4-Dinitrotoluene model/secondary 1.07e-10 1.14e-11 LPG Production Naphthalene secondary 0.79e-11 1.09e-11	US electric grid	Methyl ethyl ketone	model/secondary	2.36e-10	2.62e-11
US electric grid1,2-Dichloroethanemodel/secondary2.07e-102.30e-11US electric gridMethyl methacrylatemodel/secondary2.02e-102.24e-11US electric gridStyrenemodel/secondary1.78e-101.98e-11LCD landfillingTrichloroethyleneprimary1.72e-101.91e-11US electric gridBromoformmodel/secondary1.66e-101.84e-11US electric gridDichloromethanemodel/secondary1.64e-101.82e-11LPG ProductionCoppersecondary1.37e-101.52e-11LPG ProductionMethyl hydrazinesecondary1.33e-101.48e-11LCD incinerationTrichloroethylenesecondary1.33e-101.48e-11LCD incinerationTrichloroethylenesecondary1.32e-101.48e-11LPG ProductionChlorobenzenemodel/secondary1.32e-101.48e-11LPG Production2-Chloroacetophenonesecondary1.27e-101.41e-11LPG ProductionDimethylbenzanthracenesecondary1.25e-101.39e-11LPG ProductionBromomethanesecondary1.25e-101.39e-11LPG ProductionBromomethanesecondary1.07e-101.19e-11US electric grid2,4-Dinitrotoluenemodel/secondary1.07e-101.19e-11US electric grid2,4-Dinitrotoluenemodel/secondary1.07e-101.19e-11US electric gridAcetophenonesecondary9.79e-111.09e-11LPG Production <t< td=""><td>US electric grid</td><td>Tetrachloroethylene</td><td>model/secondary</td><td>2.33e-10</td><td>2.59e-11</td></t<>	US electric grid	Tetrachloroethylene	model/secondary	2.33e-10	2.59e-11
US electric grid Methyl methacrylate model/secondary 2.02e-10 2.24e-11 US electric grid Styrene model/secondary 1.78e-10 1.98e-11 LCD landfilling Trichloroethylene primary 1.72e-10 1.91e-11 US electric grid Bromoform model/secondary 1.66e-10 1.84e-11 US electric grid Dichloromethane model/secondary 1.64e-10 1.82e-11 LPG Production Copper secondary 1.37e-10 1.52e-11 LPG Production Methyl hydrazine secondary 1.35e-10 1.50e-11 LPG Production Silicon secondary 1.33e-10 1.48e-11 US electric grid Chlorobenzene model/secondary 1.33e-10 1.48e-11 LPG Production 2-Chloroacetophenone secondary 1.33e-10 1.48e-11 LPG Production Dimethylbenzanthracene secondary 1.27e-10 1.41e-11 LPG Production Bromomethane secondary 1.25e-10 1.39e-11 LPG Production Bromomethane secondary 1.25e-10 1.39e-11 LPG Production Bromomethane secondary 1.07e-10 1.19e-11 US electric grid 2.4-Dinitrotoluene model/secondary 1.07e-10 1.18e-11 LCD landfilling Ethylbenzene model/secondary 1.07e-10 1.18e-11 LCD landfilling Ethylbenzene model/secondary 1.07e-10 1.18e-11 LCD landfilling Ethylbenzene model/secondary 1.07e-10 1.18e-11 LPG Production Naphthalene secondary 1.07e-10 1.18e-11	US electric grid	1.2-Dichloroethane	model/secondary	2.07e-10	2.30e-11
US electric gridStyrenemodel/secondary1.78e-101.98e-11LCD landfillingTrichloroethyleneprimary1.72e-101.91e-11US electric gridBromoformmodel/secondary1.66e-101.84e-11US electric gridDichloromethanemodel/secondary1.64e-101.82e-11LPG ProductionCoppersecondary1.37e-101.52e-11LPG ProductionMethyl hydrazinesecondary1.35e-101.50e-11LPG ProductionSiliconsecondary1.33e-101.48e-11LCD incinerationTrichloroethylenesecondary1.33e-101.48e-11US electric gridChlorobenzenemodel/secondary1.33e-101.48e-11LPG ProductionDimethylbenzanthracenesecondary1.27e-101.41e-11LPG ProductionDimethylbenzanthracenesecondary1.26e-101.39e-11LPG ProductionDimethylbenzeneprimary1.07e-101.19e-11LPG ProductionBromomethanesecondary1.25e-101.39e-11LPG ProductionBromomethanesecondary1.07e-101.19e-11LPG ProductionNaphthalenesecondary1.03e-101.14e-11US electric grid2,4-Dinitrotoluenemodel/secondary1.03e-101.14e-11US electric gridAcetophenonesecondary1.03e-101.14e-11US electric gridAcetophenonesecondary1.03e-101.14e-11US electric gridAcetophenonesecondary9.79e-	US electric grid	Methyl methacrylate	model/secondary	2.02e-10	2.24e-11
LCD landfillingTrichloroethyleneprimary1.72e-101.91e-11US electric gridBromoformmodel/secondary1.66e-101.84e-11US electric gridDichloromethanemodel/secondary1.64e-101.82e-11LPG ProductionCoppersecondary1.37e-101.52e-11LPG ProductionMethyl hydrazinesecondary1.35e-101.50e-11LPG ProductionSiliconsecondary1.33e-101.48e-11LCD incinerationTrichloroethylenesecondary1.33e-101.48e-11LCD productionChlorobenzenemodel/secondary1.33e-101.48e-11LPG Production2-Chlorobenzenemodel/secondary1.32e-101.48e-11LPG ProductionDimethylbenzanthracenesecondary1.27e-101.41e-11LPG ProductionDimethylbenzanthracenesecondary1.26e-101.39e-11LCD landfillingEthylbenzeneprimary1.07e-101.19e-11US electric grid2,4-Dinitrotoluenemodel/secondary1.07e-101.19e-11US electric grid2,4-Dinitrotoluenemodel/secondary1.03e-101.14e-11US electric gridAcetophenonesecondary1.03e-101.14e-11US electric gridAcetophenonesecondary9.79e-111.09e-11US electric gridAcetophenonesecondary9.79e-111.09e-11US electric gridAcetophenonesecondary9.79e-111.09e-11US electric gridAcetophenonesec	US electric grid	Styrene	model/secondary	1.78e-10	1.98e-11
US electric gridBromoformmodel/secondary1.66e-101.84e-11US electric gridDichloromethanemodel/secondary1.64e-101.82e-11LPG ProductionCoppersecondary1.37e-101.52e-11LPG ProductionMethyl hydrazinesecondary1.35e-101.50e-11LPG ProductionSiliconsecondary1.33e-101.48e-11LCD incinerationTrichloroethylenesecondary1.33e-101.48e-11US electric gridChlorobenzenemodel/secondary1.33e-101.48e-11LPG Production2-Chloroacetophenonesecondary1.27e-101.41e-11LPG ProductionDimethylbenzanthracenesecondary1.25e-101.39e-11LPG ProductionBromomethanesecondary1.25e-101.39e-11LPG ProductionBromomethanesecondary1.07e-101.19e-11LPG ProductionBromomethanesecondary1.07e-101.19e-11LPG ProductionNaphthalenesecondary1.07e-101.19e-11US electric grid2,4-Dinitrotoluenemodel/secondary1.03e-101.14e-11US electric gridAcetophenonemodel/secondary9.79e-111.09e-11US electric gridAcetophenonemodel/secondary9.79e-111.09e-11LPG ProductionNaphthalenesecondary9.74e-111.08e-11	LCD landfilling	Trichloroethylene	primary	1.72e-10	1.91e-11
US electric gridDichloromethanemodel/secondary1.64e-101.82e-11LPG ProductionCoppersecondary1.37e-101.52e-11LPG ProductionMethyl hydrazinesecondary1.35e-101.50e-11LPG ProductionSiliconsecondary1.33e-101.48e-11LCD incinerationTrichloroethylenesecondary1.33e-101.48e-11US electric gridChlorobenzenemodel/secondary1.33e-101.48e-11LPG Production2-Chloroacetophenonesecondary1.27e-101.41e-11LPG ProductionDimethylbenzanthracenesecondary1.25e-101.39e-11LPG ProductionBromomethanesecondary1.25e-101.39e-11LPG ProductionBromomethanesecondary1.07e-101.19e-11US electric grid2,4-Dinitrotoluenemodel/secondary1.07e-101.18e-11LPG ProductionNaphthalenesecondary1.03e-101.14e-11LPG ProductionLeadsecondary1.03e-101.14e-11	US electric grid	Bromoform	model/secondary	1.66e-10	1.84e-11
LPG ProductionCoppersecondary1.37e-101.52e-11LPG ProductionMethyl hydrazinesecondary1.35e-101.50e-11LPG ProductionSiliconsecondary1.33e-101.48e-11LCD incinerationTrichloroethylenesecondary1.33e-101.48e-11US electric gridChlorobenzenemodel/secondary1.33e-101.48e-11LPG Production2-Chloroacetophenonesecondary1.27e-101.41e-11LPG ProductionDimethylbenzanthracenesecondary1.26e-101.39e-11LPG ProductionBromomethanesecondary1.25e-101.39e-11LCD landfillingEthylbenzeneprimary1.07e-101.19e-11US electric grid2,4-Dinitrotoluenemodel/secondary1.03e-101.14e-11US electric gridAcetophenonesecondary1.03e-101.14e-11LPG ProductionNaphthalenesecondary1.03e-101.14e-11US electric gridAcetophenonemodel/secondary9.79e-111.09e-11LPG ProductionNaphthalenesecondary9.79e-111.09e-11	US electric grid	Dichloromethane	model/secondary	1.64e-10	1.82e-11
LPG ProductionMethyl hydrazinesecondary1.35e-101.50e-11LPG ProductionSiliconsecondary1.33e-101.48e-11LCD incinerationTrichloroethylenesecondary1.33e-101.48e-11US electric gridChlorobenzenemodel/secondary1.33e-101.48e-11LPG Production2-Chloroacetophenonesecondary1.27e-101.41e-11LPG ProductionDimethylbenzanthracenesecondary1.26e-101.39e-11LPG ProductionBromomethanesecondary1.25e-101.39e-11LCD landfillingEthylbenzeneprimary1.07e-101.19e-11US electric grid2,4-Dinitrotoluenemodel/secondary1.03e-101.14e-11LPG ProductionNaphthalenesecondary1.03e-101.14e-11LPG ProductionLeadsecondary9.79e-111.09e-11	LPG Production	Copper	secondary	1.37e-10	1.52e-11
LPG ProductionSiliconsecondary1.33e-101.48e-11LCD incinerationTrichloroethylenesecondary1.33e-101.48e-11US electric gridChlorobenzenemodel/secondary1.33e-101.48e-11LPG Production2-Chloroacetophenonesecondary1.27e-101.41e-11LPG ProductionDimethylbenzanthracenesecondary1.26e-101.39e-11LPG ProductionBromomethanesecondary1.25e-101.39e-11LPG ProductionBromomethanesecondary1.07e-101.19e-11US electric grid2,4-Dinitrotoluenemodel/secondary1.07e-101.18e-11LPG ProductionNaphthalenesecondary1.03e-101.14e-11LPG ProductionLeadsecondary9.79e-111.09e-11	LPG Production	Methyl hydrazine	secondary	1.35e-10	1.50e-11
LCD incinerationTrichloroethylenesecondary1.33e-101.48e-11US electric gridChlorobenzenemodel/secondary1.33e-101.48e-11LPG Production2-Chloroacetophenonesecondary1.27e-101.41e-11LPG ProductionDimethylbenzanthracenesecondary1.26e-101.39e-11LPG ProductionBromomethanesecondary1.25e-101.39e-11LCD landfillingEthylbenzeneprimary1.07e-101.19e-11US electric grid2,4-Dinitrotoluenemodel/secondary1.07e-101.18e-11LPG ProductionNaphthalenesecondary1.03e-101.14e-11LPG ProductionLeadsecondary9.79e-111.09e-11	LPG Production	Silicon	secondary	1.33e-10	1.48e-11
US electric gridChlorobenzenemodel/secondary1.33e-101.48e-11LPG Production2-Chloroacetophenonesecondary1.27e-101.41e-11LPG ProductionDimethylbenzanthracenesecondary1.26e-101.39e-11LPG ProductionBromomethanesecondary1.25e-101.39e-11LCD landfillingEthylbenzeneprimary1.07e-101.19e-11US electric grid2,4-Dinitrotoluenemodel/secondary1.07e-101.18e-11LPG ProductionNaphthalenesecondary1.03e-101.14e-11LPG ProductionLeadsecondary9.79e-111.09e-11	LCD incineration	Trichloroethylene	secondary	1.33e-10	1.48e-11
LPG Production2-Chloroacetophenonesecondary1.27e-101.41e-11LPG ProductionDimethylbenzanthracenesecondary1.26e-101.39e-11LPG ProductionBromomethanesecondary1.25e-101.39e-11LCD landfillingEthylbenzeneprimary1.07e-101.19e-11US electric grid2,4-Dinitrotoluenemodel/secondary1.07e-101.18e-11LPG ProductionNaphthalenesecondary1.03e-101.14e-11LPG ProductionLeadsecondary9.79e-111.09e-11	US electric grid	Chlorobenzene	model/secondary	1 33e-10	1 48e-11
LPG ProductionDimethylbenzanthracenesecondary1.2fe-101.39e-11LPG ProductionBromomethanesecondary1.25e-101.39e-11LCD landfillingEthylbenzeneprimary1.07e-101.19e-11US electric grid2,4-Dinitrotoluenemodel/secondary1.07e-101.18e-11LPG ProductionNaphthalenesecondary1.03e-101.14e-11US electric gridAcetophenonemodel/secondary9.79e-111.09e-11LPG ProductionLeadsecondary9.74e-111.08e-11	LPG Production	2-Chloroacetonhenone	secondary	1 27e-10	1 41e-11
LPG ProductionBromomethanesecondary1.25e-101.39e-11LCD landfillingEthylbenzeneprimary1.07e-101.19e-11US electric grid2,4-Dinitrotoluenemodel/secondary1.07e-101.18e-11LPG ProductionNaphthalenesecondary1.03e-101.14e-11US electric gridAcetophenonemodel/secondary9.79e-111.09e-11LPG ProductionLeadsecondary9.74e-111.08e-11	LPG Production	Dimethylbenzanthracene	secondary	1.27e-10	1 39e-11
LCD landfillingEthylbenzeneprimary1.07e-101.19e-11US electric grid2,4-Dinitrotoluenemodel/secondary1.07e-101.18e-11LPG ProductionNaphthalenesecondary1.03e-101.14e-11US electric gridAcetophenonemodel/secondary9.79e-111.09e-11LPG ProductionLeadsecondary9.74e-111.08e-11	LPG Production	Bromomethane	secondary	1.25e-10	1.39e-11
US electric grid2,4-Dinitrotoluenemodel/secondary1.07e-101.18e-11LPG ProductionNaphthalenesecondary1.03e-101.14e-11US electric gridAcetophenonemodel/secondary9.79e-111.09e-11LPG ProductionLeadsecondary9.74e-111.08e-11	LCD landfilling	Ethylbenzene	primary	1.220 10 1.07e-10	1 19e-11
LPG ProductionNaphthalenesecondary1.0/e 101.10/e 11US electric gridAcetophenonemodel/secondary9.79e-111.09e-11LPG ProductionLeadsecondary9.74e-111.08e-11	US electric grid	2.4-Dinitrotoluene	model/secondary	1.07e-10	1.18e-11
US electric gridAcetophenonemodel/secondary9.79e-111.09e-11LPG ProductionLeadsecondary9.74e-111.08e-11	LPG Production	Naphthalene	secondary	1 03e-10	1 14e-11
LPG Production Lead secondary 974e-11 108e-11	US electric grid	Acetophenone	model/secondary	9 79e-11	1.09e-11
	LPG Production	Lead	secondary	9.74e-11	1.08e-11

 Table M-34.
 LCD LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
LPG Production	Manganese	secondary	8.79e-11	9.75e-12
LPG Production	Beryllium	secondary	8.45e-11	9.37e-12
LPG Production	Barium	secondary	7.34e-11	8.14e-12
LPG Production	Cyanide (-1)	secondary	6.59e-11	7.31e-12
LPG Production	Barium cmpds	secondary	6.25e-11	6.93e-12
US electric grid	Ethylbenzene	model/secondary	5.24e-11	5.82e-12
LCD landfilling	Dichloromethane	primary	4.84e-11	5.36e-12
LPG Production	Cadmium	secondary	3.63e-11	4.02e-12
LCD landfilling	Vinyl chloride	primary	3.62e-11	4.01e-12
LCD incineration	Vinyl chloride	secondary	2.81e-11	3.11e-12
LPG Production	Carbon disulfide	secondary	2.70e-11	2.99e-12
US electric grid	Methyl tert-butyl ether	model/secondary	2.65e-11	2.93e-12
LPG Production	Benzyl chloride	secondary	2.06e-11	2.29e-12
US electric grid	Phenol	model/secondary	2.02e-11	2.24e-12
LPG Production	Cobalt	secondary	1 98e-11	2.19e-12
US electric grid	Vinyl acetate	model/secondary	1.88e-11	2.09e-12
US electric grid	Phenanthrene	model/secondary	1.00 c 11	1.97e-12
US electric grid	Ethylene dibromide	model/secondary	1.77e-11	1.97¢ 12
US electric grid	Xylene (mixed isomers)	model/secondary	1.750 11 1.56e-11	1.73e-12
I PG Production	Aluminum (elemental)	secondary	1.50e-11	1.69e-12
LPG Production	Chloroform	secondary	1.520-11 1.46e-11	1.6)e-12
US electric grid	Chromium (III)	model/secondary	1.40e-11	1.57e-12
LPG Production	Propionaldehyde	model/secondary	1.400-11	1.350-12
LCD incincration		secondary	0.0% 12	1.496-12
LCD memeration	2.2.7.8 TODE	model/secondamy	9.060-12	0.162.12
LCD landfilling	2,5,7,6-TCDF	mimory	7.020.12	9.100-13
LCD landming	Chronnum (III)	primary	7.926-12	8.786-13
L DC Braduation	A stalain	model/secondary	7.086-12	8.526-13
LPG Production	Acrolem Mathadalahlarida	secondary	7.046-12	8.486-13
LPG Production	A seventethere	secondary	6.34e-12	7.03e-13
US electric grid	Etherl Chlarida	model/secondary	5.396-12	5.986-13
		model/secondary	5.096-12	5.05e-13
LPG Production		secondary	4.5/e-12	5.0/e-13
		model/secondary	4.51e-12	4.786-13
LPG Production	Acetaidenyde	secondary	3.64e-12	4.03e-13
LPG Production	Mercury	secondary	3.42e-12	3.80e-13
LPG Production	Cadmium empds	secondary	2.74e-12	3.04e-13
US electric grid	Biphenyl	model/secondary	2.5/e-12	2.85e-13
LPG Production	Dimethyl sulfate	secondary	2.53e-12	2.81e-13
LPG Production	loluene	secondary	2.45e-12	2.72e-13
US electric grid	Acenaphthylene	model/secondary	1.60e-12	1.77e-13
US electric grid	Benzo[a]pyrene	model/secondary	1.51e-12	1.68e-13
US electric grid	Benzo[b,j,k]fluoranthene	model/secondary	1.51e-12	1.67e-13
LPG Production	1,4-Dichlorobenzene	secondary	1.43e-12	1.58e-13
US electric grid	Anthracene	model/secondary	1.40e-12	1.55e-13
LPG Production	Isophorone	secondary	1.22e-12	1.36e-13
LPG Production	Aromatic hydrocarbons	secondary	1.10e-12	1.22e-13
US electric grid	Benzo[a]anthracene	model/secondary	1.00e-12	1.11e-13
LPG Production	Methyl ethyl ketone	secondary	9.79e-13	1.09e-13
LPG Production	Tetrachloroethylene	secondary	9.67e-13	1.07e-13
LPG Production	1,2-Dichloroethane	secondary	8.60e-13	9.54e-14
LPG Production	Methyl methacrylate	secondary	8.36e-13	9.28e-14

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
LPG Production	Styrene	secondary	7.39e-13	8.20e-14
US electric grid	Chrysene	model/secondary	7.27e-13	8.07e-14
US electric grid	Indeno(1,2,3-cd)pyrene	model/secondary	7.13e-13	7.91e-14
LPG Production	Bromoform	secondary	6.90e-13	7.66e-14
LPG Production	Dichloromethane	secondary	6.81e-13	7.55e-14
LPG Production	Chromium (III)	secondary	6.76e-13	7.50e-14
US electric grid	Fluorene	model/secondary	5.66e-13	6.28e-14
LPG Production	Chlorobenzene	secondary	5.52e-13	6.12e-14
LPG Production	Copper (+1 & +2)	secondary	5.16e-13	5.72e-14
US electric grid	Fluoranthene	model/secondary	4.52e-13	5.02e-14
LPG Production	2,4-Dinitrotoluene	secondary	4.43e-13	4.91e-14
LPG Production	Acetophenone	secondary	4.06e-13	4.51e-14
US electric grid	Dibenzo[a,h]anthracene	model/secondary	3.88e-13	4.31e-14
US electric grid	2-Methylnaphthalene	model/secondary	3.78e-13	4.19e-14
US electric grid	Pyrene	model/secondary	3.74e-13	4.15e-14
US electric grid	5-Methyl chrysene	model/secondary	2.79e-13	3.10e-14
LPG Production	o-xvlene	secondary	2.70e-13	3.00e-14
US electric grid	o-xvlene	model/secondary	2.69e-13	2.98e-14
US electric grid	Benzo[g.h.i]pervlene	model/secondary	2.55e-13	2.83e-14
LPG Production	Ethylbenzene	secondary	2.19e-13	2.43e-14
LPG Production	Methyl tert-butyl ether	secondary	1 10e-13	1 22e-14
LPG Production	Zinc (+2)	secondary	9.24e-14	1.03e-14
LPG Production	Phenanthrene	secondary	8 93e-14	9 90e-15
LPG Production	Vinvl acetate	secondary	7.82e-14	8.67e-15
LPG Production	3-Methylcholanthrene	secondary	7 56e-14	8 38e-15
LPG Production	Ethylene dibromide	secondary	7.25e-14	8.04e-15
LPG Production	2-Methylnaphthalene	secondary	4 73e-14	5 25e-15
LPG Production	Chlorine	secondary	4 19e-14	4 65e-15
LPG Production	1 1 1-Trichloroethane	secondary	2 98e-14	3 31e-15
LPG Production	Acenaphthene	secondary	2.50e 11 2.77e-14	3.07e-15
LPG Production	Ethyl Chloride	secondary	2.17e-14	2 34e-15
LPG Production	Cumene	secondary	1 79e-14	1.98e-15
LPG Production	Benzo[a]nyrene	secondary	1.79e-14	1.90e 15
LPG Production	Binhenvl	secondary	1.07e-14	1.070 15
LPG Production	L ead cmnds	secondary	9.19e-15	1.10e 15
LPG Production	Dibenzo[a h]anthracene	secondary	8 77e-15	9.72e-16
LPG Production	Acenanhthylene	secondary	8.43e-15	9.35e-16
LPG Production	Anthracene	secondary	8 31e-15	9.22e-16
LPG Production	Nickel cmpds	secondary	7 77e-15	8.62e-16
LPG Production	Benzo[a]anthracene	secondary	7.516-15	8 33e-16
LPG Production	Benzo[h i k]fluoranthene	secondary	5.80e-15	6.43e-16
LPG Production	Indeno(1.2.3. cd)pyrene	secondary	5 50e 15	6 20e 16
LPG Production	Chrysene	secondary	4.74e-15	5.26e-16
LPG Production	HALON 1301	secondary	4.00e.15	1 110 16
L PG Production	Renzo[h]fluorenthene	secondary	4.000-15	3 220 16
L PG Production	Mercury compounds	secondary	2 70 - 15	3.00-16
L PG Production	Fluorene	secondary	2.798-15	2 99- 16
L DG Production	Durana	secondary	2.000-15	2.000-10
L DG Production	Ponzola h ilnomilona	secondary	2.236-15	2.490-10
	Denzolg, n, i perviene	secondary	2.196-15	2.420-10

 Table M-34.
 LCD LCIA Results for the Chronic Public Health Effects Impact Category

LPG Production Fluoranthene secondary 1.12-15 2.35-16 LPG Production Halogenated matter (organic) secondary 9.10e-16 1.02e-16 LPG Production Halogenated hydrocarbons (unspecified) secondary 2.38e-18 2.55e-19 LPG Induction Halogenated hydrocarbons (unspecified) secondary -3.36e-18 -3.3ee-19 LCD inducation Halogenated hydrocarbons (unspecified) secondary -4.34e-16 -1.65e-16 Natural Gas Prod. Halogenated hydrocarbons (unspecified) secondary -4.74e-15 -5.25e-16 Natural Gas Prod. Mattor (Cas Prod. Mattor (Cas Prod. -1.75e-16 Natural Gas Prod. -1.75e-16 Natural Gas Prod. Nickel (corganic) secondary -1.35e-14 -1.46e-15 Natural Gas Prod. HALON-1001 secondary -1.35e-14 -1.46e-13 LCD incineration HALON-1001 secondary -0.46e-13 -1.85e-14 LCD incineration HALON-1001 secondary -0.46e-13 -1.85e-14 LCD incineration Nickel (corganic) <t< th=""><th>Process Group</th><th>Material</th><th>LCI Data Type</th><th>Chronic Public</th><th>% of Total</th></t<>	Process Group	Material	LCI Data Type	Chronic Public	% of Total
EPG Production 5-Methyl chrysene secondary 116e-15 129e-16 LPG Production Halogemetel matter (organic) secondary 9.19e-16 1.02e-16 LPG Production Halogemetel hydrocarbots (mspecified) secondary 4.39e-18 4.35e-19 Natural Gas Prod. Halogemetel hydrocarbots (mspecified) secondary 4.48e-16 4.58e-18 LO incinentical Halogemetel hydrocarbots (mspecified) secondary 4.46e-16 4.58e-18 Natural Gas Prod. Halogemetel hydrocarbots (mspecified) secondary 4.74e-15 4.52e-16 Natural Gas Prod. Halogemetel hydrocarbots (mspecified) secondary 4.76e-15 4.73e-16 Natural Gas Prod. Halogemetel hydrocarbots (mspecified) secondary 4.78e-15 4.46e-15 LCD incinention Halogemetel hydrocarbots (mspecified) secondary 4.78e-13 4.78e-14 LCD incinention Halogemetel hydrocarbots (mspecified) secondary 4.78e-13 4.78e-13 LCD incinention Halogemetel hydrocarbots (mspecified) secondary 4.78e-13 4.78e-13 LCD	LPG Production	Fluoranthene	secondary	2 12e-15	2 35e-16
LPG ProductionHalogenated matter (organic)secondary9.19e.161.02e.16LPG ProductionHalogenated matter (organic)secondary2.30e.182.258e.19Natural Gar Prod.Halogenated hydrocarbons (unspecified)secondary3.30e.182.258e.19LCD incinerationHalogenated hydrocarbons (unspecified)secondary $-9.44e.16$ $-1.03e.16$ Natural Gar Prod.Halogenated hydrocarbons (unspecified)secondary $-9.44e.16$ $-1.03e.16$ Natural Gar Prod.Mercury compoundssecondary $-1.56e.15$ $-7.54e.16$ Natural Gar Prod.Mercury compoundssecondary $-1.56e.15$ $-5.75e.16$ Natural Gar Prod.HALON-1301secondary $-1.32e.14$ $-1.44e.15$ Natural Gar Prod.Halogenated matter (organic)secondary $-1.08e.14$ $-2.38e.14$ $-2.64e.15$ LCD incinerationHalogenated matter (organic)secondary $-1.08e.13$ $-1.18e.14$ LCD incinerationHalogenated matter (organic)secondary $-4.38e.14$ $-2.64e.15$ LCD incinerationNickel empdssecondary $-4.38e.13$ $-3.38e.14$ Natural Gas Prod.Fluoranthenesecondary $-4.38e.13$ $-3.38e.14$ Valid Gas Prod.Fluoranthenesecondary $-5.48e.13$ $-7.30e.14$ LCD incinerationRozaplifiloranthenesecondary $-5.8e.13$ $-7.30e.14$ LCD incinerationBerzoplifiloranthenesecondary $-5.8e.13$ $-7.30e.14$ LCD GineinstrationBerzoplif	LPG Production	5-Methyl chrysene	secondary	1 16e-15	1 29e-16
LPG Production Halogenated hydroxarbos (unspecified) secondary 2.30e-18 2.58e-10 Natural Gas Prod. Halogenated hydroxarbos (unspecified) secondary 4.318-10 4.338-10 LCD incincention Halogenated hydroxarbos (unspecified) secondary 4.318-16 1.105e-16 Natural Gas Prod. Halogenated hydroxarbos (unspecified) secondary 4.74e-15 5.52e-16 Natural Gas Prod. Halogenated matter (organic) secondary 4.74e-15 5.52e-16 Natural Gas Prod. HALON-1301 secondary -1.32e-14 -1.47e-15 Natural Gas Prod. Halogenated matter (organic) secondary -1.32e-14 -1.47e-15 Natural Gas Prod. Halogenated matter (organic) secondary -1.38e-14 -1.47e-15 Natural Gas Prod. Halogenated matter (organic) secondary -1.38e-14 -1.47e-16 Natural Gas Prod. Halogenated matter (organic) secondary -1.38e-13 -1.02e-14 LCD incincention Halogenated matter (organic) secondary -1.68e-13 -1.37e-14 LCD incincention	LPG Production	Halogenated matter (organic)	secondary	9 19e-16	1.02e-16
Jack Hall Indegenated hydrocarbons (unspecified) secondary 3.90e1.8 4.33e1.9 LCD incinentation Halogenated hydrocarbons (unspecified) secondary 4.13e1.7 4.43e1.9 LCD information Halogenated hydrocarbons (unspecified) secondary 9.44e1.6 1.05e.16 Natural Gas Prod. Halogenated hydrocarbons (unspecified) secondary 4.76e1.5 5.75e.16 Natural Gas Prod. HALON-1301 secondary 4.76e1.5 5.75e.16 Natural Gas Prod. Halogenated matter (organic) secondary 4.76e1.5 4.75e.15 LCD incineration Halogenated matter (organic) secondary 4.76e1.5 4.75e.16 LCD incineration Halogenated matter (organic) secondary 4.76e.13 4.18e.14 LCD incineration Nickel empth secondary 4.78e.13 4.738e.14 LCD incineration Nickel empth secondary 4.78e.13 4.738e.14 LCD incineration Rotoration secondary 4.88e.13 4.538e.14 LCD incineration Bacorghyhiperyten secondary <t< td=""><td>I PG Production</td><td>Halogenated hydrocarbons (unspecified)</td><td>secondary</td><td>2 30e-18</td><td>2 55e-19</td></t<>	I PG Production	Halogenated hydrocarbons (unspecified)	secondary	2 30e-18	2 55e-19
LCD indimination Halogenated hydroarbors (unspecified) secondary 4.138 ± 17 4.38 ± 17 Fuel 01 H 4 Prod. Halogenated hydroarbors (unspecified) secondary -4.138 ± 17 -4.38 ± 18 Natural Gas Prod. Halogenated matter (organic) secondary $-7.78 \pm 17.83 \pm 10.137	Natural Gas Prod	Halogenated hydrocarbons (unspecified)	secondary	-3 90e-18	-4 33e-19
End Oil # Prod. Halogenated hydroarbox (unspecified) secondary 4.34-16 4.108-16 Natural Gas Prod. Halogenated indroarbox (unspecified) secondary 4.73e-15 5.23e-16 Natural Gas Prod. HALON-1301 secondary 4.73e-15 5.23e-16 Natural Gas Prod. Lead empds secondary 4.73e-15 5.23e-14 1.14e-15 Natural Gas Prod. Lead empds secondary 4.13e-14 1.178e-16 LCD incineration Halogenated matre (organic) secondary 4.13e-14 1.178e-16 LCD incineration Halogenated matre (organic) secondary 4.13e-14 1.178e-16 Natural Gas Prod. Halogenated matre (organic) secondary 4.18e-14 1.46e-15 Valuer Gas Prod. Halogenated matre (organic) secondary 4.38e-13 4.19e-14 Natural Gas Prod. Halogenated matre (organic) secondary 4.38e-13 4.19e-14 Natural Gas Prod. Fluoranthene secondary 4.38e-13 4.39e-14 LCD incineration BenzolpHubaranthene secondary	LCD incineration	Halogenated hydrocarbons (unspecified)	secondary	-4 13e-17	-4 58e-18
Natural Gas Prod.Halogenate matter (arganic)secondary4.136e-131.178e-16Natural Gas Prod.HALON-1301secondary-6.79e-157.754e-16Natural Gas Prod.Lead empdssecondary-1.32e-14-1.46e-15Natural Gas Prod.Lead empdssecondary-2.38e-14-2.264e-15LCD incincerationHaLON-1301secondary-2.38e-14-2.264e-15LCD incincerationHalogenatel matter (organic)secondary-1.08e-13-1.88e-14LCD incincerationNickel empdssecondary-3.78e-13-2.22e-14LCD incincerationNickel empdssecondary-3.78e-13-2.22e-14LCD incincerationNickel empdssecondary-3.78e-13-2.22e-14LCD incincerationNickel empdssecondary-3.78e-13-4.19e-14LCD incincerationNickel empdssecondary-4.8e-13-5.38e-14LCD incincerationBenzolpHoranthenesecondary-6.8e-13-7.76e-14Natural Gas Prod.Fluorenesecondary-6.8e-13-7.76e-14Fuel Oil 44 Prod.S-Methyl chrysenesecondary-8.8e-13-7.76e-14Natural Gas Prod.BenzolpLik/Horanthenesecondary-8.8e-13-7.76e-14Natural Gas Prod.BenzolpLik/Horanthenesecondary-1.8e-13-1.78e-14Natural Gas Prod.BenzolpLik/Horanthenesecondary-1.18e-12-1.17e-13Pael Oil 44 Prod.BenzolpLik/Horanthenesecondary-1.18e-12-1.17e-13 </td <td>Fuel Oil #4 Prod</td> <td>Halogenated hydrocarbons (unspecified)</td> <td>secondary</td> <td>-9.44e-16</td> <td>-1.05e-16</td>	Fuel Oil #4 Prod	Halogenated hydrocarbons (unspecified)	secondary	-9.44e-16	-1.05e-16
Journal Gas Prod.Integration (segurity)Journal (Segurity)Journal (Jas Prod.Journal (Jas Prod.Jo	Natural Gas Prod	Halogenated matter (organic)	secondary	-1 56e-15	-1 73e-16
Natural Gas Prod.HALON-1301secondary6.79e-157.25e-16Natural Gas Prod.Nickel empdssecondary-1.35e-14-1.73e-15LCD incincrationHalogenated matter (organic)secondary-1.36e-14-1.73e-15LCD incincrationHalogenated matter (organic)secondary-1.36e-14-1.73e-15LCD incincrationHalogenated matter (organic)secondary-1.09e-13-1.88e-14LCD incincrationNickel empdssecondary-2.01e-13-2.23e-14LCD incincrationNickel empdssecondary-3.78e-13-4.19e-14LCD incincrationNickel empdssecondary-3.78e-13-4.19e-14Natural Gas Prod.Fluoranthenesecondary-5.34e-13-5.38e-14Natural Gas Prod.Fluoranthenesecondary-6.98e-13-7.76e-14Valural Gas Prod.Pyrenesecondary-6.98e-13-7.76e-14Natural Gas Prod.Pyrenesecondary-8.45e-13-0.38e-14Natural Gas Prod.Pyrenesecondary-8.15e-13-0.38e-14Natural Gas Prod.Pyrenesecondary-1.15e-12-1.72e-13Puel Oil #4 Prod.Benzo[f], k] fluoranthenesecondary-1.48e-13-9.38e-14Natural Gas Prod.Copper (+1.&=2)secondary-1.48e-12-1.27e-13Puel Oil #4 Prod.Benzo[f], k] pery lenesecondary-1.36e-12-1.72e-13Puel Oil #4 Prod.Benzo[f], h] pery lenesecondary-1.48e-12-1.26e-12-1.27e-13	Natural Gas Prod	Mercury compounds	secondary	-4 74e-15	-5 25e-16
JanuardDiractor (Jacobia)Diractor (Jacobia)Natural Gas Prod.Lead empdssecondary-1.32e+14LCD incinerationHalogenated matter (organic)secondary-2.38e+14LCD incinerationHALON-1301secondary-2.38e+14Natural Gas Prod.5-Methyl chrysnesecondary-2.01e+13LCD incinerationNickel empdssecondary-2.01e+13-2.25e+14Natural Gas Prod.Fluoranthenesecondary-3.78e+13-4.19e+14Natural Gas Prod.Fluoranthenesecondary-4.58e+13-5.38e+14LCD incinerationBerzo(b)fluoranthenesecondary-5.58e+13-5.38e+14LCD incinerationBerzo(b)fluoranthenesecondary-6.58e+13-5.38e+14Natural Gas Prod.Pyrenesecondary-6.58e+13-5.38e+14Natural Gas Prod.Pyrenesecondary-8.10e+13-8.99e+14Natural Gas Prod.Pyrenesecondary-8.10e+13-8.99e+14Natural Gas Prod.Pyrenesecondary-1.15e+12-1.17e+13Fuel OII #4 Prod.Berzo(g), J.J[huoranthenesecondary-1.15e+12-1.17e+13Fuel OII #4 Prod.Berzo(g), J.J[pyrelnesecondary-1.15e+12-1.17e+13Fuel OII #4 Prod.Berzo(g), J.J[pyrelnesecondary-1.15e+12-1.27e+13Fuel OII #4 Prod.Berzo(g), J.J[pyrelnesecondary-1.15e+12-1.27e+13Fuel OII #4 Prod.Berzo(g), J.J[pyrelnesecondary-1.26e+12-1.17e+13 <td>Natural Gas Prod</td> <td>HALON-1301</td> <td>secondary</td> <td>-6 79e-15</td> <td>-7 54e-16</td>	Natural Gas Prod	HALON-1301	secondary	-6 79e-15	-7 54e-16
$\begin{tabular}{lllllllllllllllllllllllllllllllllll$	Natural Gas Prod	Nickel cmpds	secondary	-0.77C-15	-1.34c-15
NatureConstructionHalogenized matter (organic)secondary $-2.38-14$ $-2.54-15$ LCD incinerationHALON-1301secondary $-1.036-13$ $-1.15e-14$ LCD incinerationNickel empdssecondary $-1.69e-13$ $-1.88e-14$ LCD incinerationNickel empdssecondary $-2.38e+14$ $-2.38e+14$ LCD incinerationNickel empdssecondary $-3.78e-13$ $-4.19e-14$ LCD incinerationNickel empdssecondary $-4.85e+13$ $-5.38e+14$ LCD incinerationBenzolb/Horanthenesecondary $-4.85e+13$ $-5.38e+14$ LCD incinerationBenzolb/Horanthenesecondary $-4.85e+13$ $-6.02e+14$ LCD incinerationBenzolb/Horanthenesecondary $-4.59e+13$ $-7.78e+14$ LCD incinerationBenzolb/Horanthenesecondary $-4.59e+13$ $-7.30e+14$ LCD incinerationBenzolb/Horanthenesecondary $-8.7e+13$ $-9.38e+14$ Natural Gas Prod.Benzolb/Horanthenesecondary $-4.59e+13$ $-7.2e+14$ Natural Gas Prod.Benzolb/Horanthenesecondary $-1.15e+12$ $-1.27e+13$ Fuel Oil# Prod.Benzolb/Horanthenesecondary $-1.15e+12$ $-1.27e+13$ Fuel Oil# Prod.Benzolb/Horanthenesecondary $-1.26e+12$ $-1.78e+14$ Fuel Oil# Prod.Pronesecondary $-1.26e+12$ $-1.78e+13$ Fuel Oil# Prod.Promesecondary $-1.26e+12$ $-1.78e+13$ Fuel Oil# Prod.Promeseconda	Natural Gas Prod	Lead empds	secondary	-1.520-14	-1.73e-15
Leb indirentionHalogenact matter (nganc)secondary $-1.08-13$ $-1.58-14$ Natural Gas Prod.5-Methyl chryseresecondary $-1.08-13$ $-1.58-14$ LCD incinerationNickel empdssecondary $-2.01e-13$ $-2.23e-14$ LCD IndirentationHalogenated matter (organic)secondary $-3.78e-13$ $-4.19e-14$ Natural Gas Prod.Fluoranthenesecondary $-4.85e-13$ $-5.38e-14$ LCD incinerationBenzo[b]fluoranthenesecondary $-6.58e-13$ $-7.30e-14$ LCD incinerationBenzo[b]fluoranthenesecondary $-6.58e-13$ $-7.30e-14$ Natural Gas Prod.Pyrenesecondary $-8.45e-13$ $-9.38e-14$ Natural Gas Prod.Benzo[b,jk]fluoranthenesecondary $-8.45e-13$ $-9.38e-14$ Natural Gas Prod.Benzo[b,jk]fluoranthenesecondary $-8.45e-13$ $-9.38e-14$ Natural Gas Prod.Benzo[b,jk]fluoranthenesecondary $-1.15e-12$ $-1.72e-13$ Fuel Oil #4 Prod.Benzo[b,jk]fluoranthenesecondary $-1.15e-12$ $-1.72e-13$ Fuel Oil #4 Prod.Benzo[b,jk]fluoranthenesecondary $-1.15e-12$ $-1.72e-13$ Fuel Oil #4 Prod.Benzo[b,jk]fluoranthenesecondary $-1.26e-12$ $-1.40e-13$ Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary $-1.26e-12$ $-1.40e-13$ Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary $-1.26e-12$ $-1.72e-13$ Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary $-1.26e-12$ <td>L CD incincration</td> <td>Helegeneted matter (organic)</td> <td>secondary</td> <td>-1.300-14</td> <td>-1.736-15</td>	L CD incincration	Helegeneted matter (organic)	secondary	-1.300-14	-1.736-15
LCD indicationIFADOVE1501Secondary1.68e-131.18e-14LCD incinerationNickel empdssecondary2.01e-132.23e-14Fuel Oil #4 Prod.Halogenated matter (organic)secondary3.78e-134.19e-14Natural Gas Prod.Fluorenthenesecondary4.85e-135.38e-13LCD incinerationBenzo[b]fluoranthenesecondary6.58e-137.70e-14LCD incinerationBenzo[b]fluoranthenesecondary6.58e-137.70e-14Patel Oil #4 Prod.5.Metpl) chrysenesecondary6.59e-137.70e-14Natural Gas Prod.Pyrenesecondary8.45e-139.938e-14Natural Gas Prod.Copper (+1 & +2)secondary8.45e-139.938e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.15e-12-1.17e-13Fuel Oil #4 Prod.Mercury compoundssecondary-1.15e-12-1.27e-13Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.24e-12-1.38e-13Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.24e-12-1.38e-13Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.24e-12-1.47e-13Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.24e-12-1.47e-13Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.24e-12-1.47e-13Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.26e-12-1.47e-13Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.26e-12 <td< td=""><td>LCD incineration</td><td>HALON 1201</td><td>secondary</td><td>-2.386-14</td><td>-2.04e-13</td></td<>	LCD incineration	HALON 1201	secondary	-2.386-14	-2.04e-13
Natural Cas Prod.Secondary-1.08-13-1.88-14LCD incinerationNické (empdssecondary-2.01e-13-2.32e-14Natural Gas Prod.Fluoranthenesecondary-4.85e-13-5.38e-14Natural Gas Prod.Fluorenesecondary-5.43e-13-6.02e-14LCD incinerationBerzo[b]fluoranthenesecondary-6.58e-13-7.70e-14Natural Gas Prod.Pyrenesecondary-6.99e-13-7.76e-14Natural Gas Prod.Pyrenesecondary-8.10e-13-9.72e-14Natural Gas Prod.Copper (+1 & +2)secondary-8.45e-13-9.72e-14Natural Gas Prod.Benzo[b,i,j]prylenesecondary-1.17e-13-9.72e-14Natural Gas Prod.Benzo[b,i,j]prylenesecondary-1.15e-12-1.27e-13Fuel Oil #4 Prod.Benzo[b,i,j]prylenesecondary-1.15e-12-1.27e-13Fuel Oil #4 Prod.Benzo[b,i,j]prylenesecondary-1.26e-12-1.48e-13Fuel Oil #4 Prod.Benzo[b,i,j]prylenesecondary-1.26e-12-1.48e-13Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.26e-12-1.48e-13Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.26e-12-1.48e-13Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.26e-12-1.48e-13Fuel Oil #4 Prod.Biphenylsecondary-1.26e-12-1.48e-13Fuel Oil #4 Prod.Biphenylsecondary-1.26e-12-1.72e-13Fuel Oil #4 Prod.HA	Network Cas Bred	Mathyi abuyana	secondary	-1.05e-15	-1.130-14
LCD incinetationNicket impossecondary $-2.012+13$ $-2.202+13$ $-2.202+13$ Natural Gas Prod.Fluoranthenesecondary $-3.78c-13$ $-4.19c-14$ Natural Gas Prod.Fluorenesecondary $-5.43c-13$ $-6.02c-14$ Fuel Oil # A Prod.BenzolpHuoranthenesecondary $-6.58c-13$ $-7.30c-14$ Fuel Oil # A Prod.S-Methyl chrysenesecondary $-6.98c-13$ $-7.76c-14$ Natural Gas Prod.BenzolpLik/fluoranthenesecondary $-8.10c-13$ $-8.99c-14$ Natural Gas Prod.BenzolpLik/fluoranthenesecondary $-8.16c-12$ $-1.17c-13$ Puel Oil # Prod.BenzolpLik/fluoranthenesecondary $-1.15c-12$ $-1.27c-13$ Fuel Oil # Prod.BenzolpLik/fluoranthenesecondary $-1.16c-12$ $-1.17c-13$ Fuel Oil # Prod.BenzolpLifuoranthenesecondary $-1.26c-12$ $-1.40c-13$ Fuel Oil # Prod.BenzolpLifuoranthenesecondary $-1.26c-12$ $-1.40c-13$ Fuel Oil # Prod.BenzolpLifuoranthenesecondary $-1.26c-12$ $-1.70c-13$ Fuel Oil # A Prod.BenzolpLifuoranthenesecondary $-1.26c-12$ $-1.70c-13$ Fuel Oil # A Prod.HALON-1301secondary $-1.26c-12$	L CD in singration	S-Methyl chrysene	secondary	-1.096-13	-1.000-14
Tuel OII # Prod.Introgenate matter (organic)secondary $-5.78-13$ $-4.78-13$ Natural Gas Prod.Fluorenesecondary $-4.78-13$ $-6.02e-14$ LCD incinerationBenzol[b]fluoranthenesecondary $-6.58e-13$ $-7.73e-14$ Natural Gas Prod.Pyrenesecondary $-6.98e-13$ $-7.76e-14$ Natural Gas Prod.Pyrenesecondary $-8.45e-13$ $-9.38e-14$ Natural Gas Prod.Benzol[b,i,f]toranthenesecondary $-8.45e-13$ $-9.38e-14$ Natural Gas Prod.Benzol[b,i,f]perylenesecondary $-1.16e-12$ $-1.17e-13$ Fuel OII # Prod.Benzol[b,i,f]perylenesecondary $-1.16e-12$ $-1.17e-13$ Fuel OII # 4 Prod.Benzol[b,i,f]perylenesecondary $-1.16e-12$ $-1.17e-13$ Fuel OII # 4 Prod.Benzol[b,i,f]perylenesecondary $-1.26e-12$ $-1.40e-13$ Fuel OII # 4 Prod.Fluoranthenesecondary $-1.26e-12$ $-1.40e-13$ Fuel OII # 4 Prod.Pyrenesecondary $-1.26e-12$ $-1.40e-13$ Fuel OII # 4 Prod.Fluorenesecondary $-1.26e-12$ $-1.66e-12$ Natural Gas Prod.Chrysenesecondary $-1.26e-12$ $-1.82e-13$ Natural Gas Prod.Chrysenesecondary <t< td=""><td>ECD incineration</td><td>Nickel cmpds</td><td>secondary</td><td>-2.01e-13</td><td>-2.23e-14</td></t<>	ECD incineration	Nickel cmpds	secondary	-2.01e-13	-2.23e-14
Natural Gas Prod.Futoranneesecondary-4.35e-13-5.38e-14Natural Gas Prod.Fluorenesecondary-6.58e-13-7.30e-14LCD incinerationBenzo[b]fluoranthenesecondary-6.58e-13-7.30e-14Fuel Oli f 4P rod.5-Methyl chry senesecondary-8.45e-13-9.38e-14Natural Gas Prod.Benzo[b,i,k]fluoranthenesecondary-8.45e-13-9.38e-14Natural Gas Prod.Benzo[b,i,k]fluoranthenesecondary-8.45e-13-9.38e-14Natural Gas Prod.Benzo[b,i,k]fluoranthenesecondary-1.15e-12-1.17e-13Fuel Oli #4 Prod.Benzo[g,h,i]perylenesecondary-1.15e-12-1.27e-13Fuel Oli #4 Prod.Benzo[g,h,i]perylenesecondary-1.15e-12-1.27e-13Fuel Oli #4 Prod.Fluoranthenesecondary-1.15e-12-1.27e-13Fuel Oli #4 Prod.Pyrenesecondary-1.15e-12-1.27e-13Fuel Oli #4 Prod.Pyrenesecondary-1.5e-12-1.27e-13Fuel Oli #4 Prod.Benzo[b]huoranthenesecondary-1.5e-12-1.27e-13Fuel Oli #4 Prod.Benzo[b]huoranthenesecondary-1.5e-12-1.70e-13Natural Gas Prod.Benzo[b]huoranthenesecondary-1.5e-12-1.70e-13Natural Gas Prod.Chrysenesecondary-1.5e-12-1.70e-13Natural Gas Prod.Chrysenesecondary-1.5e-12-1.70e-13Natural Gas Prod.Chrysenesecondary-1.82e-12-1.82e-12 <td< td=""><td>Fuel Oil #4 Prod.</td><td>Halogenated matter (organic)</td><td>secondary</td><td>-3./8e-13</td><td>-4.19e-14</td></td<>	Fuel Oil #4 Prod.	Halogenated matter (organic)	secondary	-3./8e-13	-4.19e-14
Natural Gas Prod.Fluorenesecondary-5.45e-13-6.02c-14LCD incinerationBerzo[b]tuoranthenesecondary-6.58e-13-7.76e-14Yeuel Oil #4 Prod.5-Methyl chrysenesecondary-8.10e-13-8.99e-13-7.76e-14Natural Gas Prod.Dyrenesecondary-8.45e-13-9.38e-14Natural Gas Prod.Copper (+1 & +2)secondary-8.47e-13-9.72e-14Natural Gas Prod.Benzo[g,h,i]perylenesecondary-1.15e-12-1.17e-13Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.15e-12-1.27e-13Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.26e-12-1.40e-13Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.26e-12-1.40e-13Fuel Oil #4 Prod.Benzo[b]tuoranthenesecondary-1.26e-12-1.40e-13Fuel Oil #4 Prod.Benzo[b]tuoranthenesecondary-1.55e-12-1.70e-13Natural Gas Prod.Fluorenesecondary-1.55e-12-1.70e-13Natural Gas Prod.Benzo[b]tuoranthenesecondary-1.58e-12-1.70e-13Natural Gas Prod.HALON-1301secondary-1.64e-12-1.82e-13Natural Gas Prod.Chrysenesecondary-1.80e-12-1.99e-13LCD incinerationDibenzo[a,h]anthracenesecondary-2.32e-12-2.02e-13Natural Gas Prod.Acenapthtylenesecondary-2.32e-12-2.27e-13Natural Gas Prod.Acenapthtylenesecondary-2.32e-12<	Natural Gas Prod.	Fluorantnene	secondary	-4.85e-13	-5.38e-14
LCD memerationBerozolp/litoranthenesecondary-6.58e-13-7.30e-14Natural Gas Prod.S-Methyl chrysenesecondary-6.58e-13-9.33e-14Natural Gas Prod.Benzolb,j,k/lluoranthenesecondary-8.45e-13-9.33e-14Natural Gas Prod.Benzolb,j,k/lluoranthenesecondary-8.77e-13-9.72e-14Natural Gas Prod.Benzolg,h,i/perylenesecondary-1.15e-12-1.17e-13Fuel Oil #4 Prod.Benzolg,h,i/perylenesecondary-1.15e-12-1.27e-13Fuel Oil #4 Prod.Benzolg,h,i/perylenesecondary-1.15e-12-1.38e-13Fuel Oil #4 Prod.Pyrenesecondary-1.26e-12-1.40e-13Fuel Oil #4 Prod.Pyrenesecondary-1.55e-12-1.70e-13Fuel Oil #4 Prod.Benzolp,hi/perylenesecondary-1.55e-12-1.70e-13Fuel Oil #4 Prod.Benzolp,hi/perylenesecondary-1.55e-12-1.70e-13Fuel Oil #4 Prod.Benzolp,hi/perylenesecondary-1.55e-12-1.70e-13Natural Gas Prod.Biphenylsecondary-1.55e-12-1.70e-13Natural Gas Prod.Chrysenesecondary-1.55e-12-1.70e-13Natural Gas Prod.Chrysenesecondary-1.58e-12-1.82e-13Natural Gas Prod.Acomatic hydrocarbonssecondary-1.82e-12-2.08e-13Natural Gas Prod.Benzolp,hi/penesecondary-2.18e-12-2.08e-13Natural Gas Prod.Genratichydrocarbonssecondary-2.61e-12-	Natural Gas Prod.	Fluorene	secondary	-5.43e-13	-6.02e-14
Fuel OII #4 Prod. 5-Methy Ichrysene secondary -6.99e-13 -7.76e-14 Natural Gas Prod. Pyrene secondary -8.10e-13 -8.99e-13 -7.76e-14 Natural Gas Prod. Copper (+1 & +2) secondary -8.77e-13 -9.72e-14 Natural Gas Prod. Benzo[g,h,i]perylene secondary -1.16e-12 -1.17e-13 Fuel Oil #4 Prod. Benzo[g,h,i]perylene secondary -1.15e-12 -1.27e-13 Fuel Oil #4 Prod. Benzo[g,h,i]perylene secondary -1.16e-12 -1.27e-13 Fuel Oil #4 Prod. Benzo[g,h,i]perylene secondary -1.26e-12 -1.27e-13 Fuel Oil #4 Prod. Benzo[b]fluoranthene secondary -1.26e-12 -1.40e-13 Fuel Oil #4 Prod. Benzo[b]fluoranthene secondary -1.26e-12 -1.27e-13 Fuel Oil #4 Prod. Benzo[b]fluoranthene secondary -1.49e-13 -1.66e-13 Fuel Oil #4 Prod. Benzo[b]fluoranthene secondary -1.58e-12 -1.72e-13 Natural Gas Prod. Chrysene secondary -1.64e-12 <td< td=""><td>LCD incineration</td><td>Benzo[b]fluoranthene</td><td>secondary</td><td>-6.58e-13</td><td>-7.30e-14</td></td<>	LCD incineration	Benzo[b]fluoranthene	secondary	-6.58e-13	-7.30e-14
Natural Gas Prod.Pyrenesecondary-8.10e-13-8.39e-14Natural Gas Prod.Benzo[b,j,k]fluoranthenesecondary-8.37e-13-9.33e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.06e-12-1.17e-13Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.15e-12-1.27e-13Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.12e-12-1.38e-13Fuel Oil #4 Prod.Fluoranthenesecondary-1.24e-12-1.38e-13Fuel Oil #4 Prod.Pyrenesecondary-1.24e-12-1.38e-13Fuel Oil #4 Prod.Benzo[g,h]perylenesecondary-1.24e-12-1.66e-13Fuel Oil #4 Prod.Fluoranthenesecondary-1.55e-12-1.70e-13Natural Gas Prod.Biphenylsecondary-1.55e-12-1.72e-13Fuel Oil #4 Prod.Fluorenesecondary-1.52e-12-1.32e-13Natural Gas Prod.Chrysenesecondary-1.82e-12-2.08e-13Natural Gas Prod.Chrysenesecondary-1.87e-12-2.02e-13Natural Gas Prod.Chrysenesecondary-2.11e-12-2.34e-13Natural Gas Prod.Denzo[b]fluoranthenesecondary-2.12e-12-2.34e-13Natural Gas Prod.Acomatic hydrocarbonssecondary-2.12e-12-2.26e-12Natural Gas Prod.Acomatic hydrocarbonssecondary-2.12e-12-2.27e-13Natural Gas Prod.Chrysenesecondary-2.21e-12-2.28e-13Natural Gas Prod. <td>Fuel Oil #4 Prod.</td> <td>5-Methyl chrysene</td> <td>secondary</td> <td>-6.99e-13</td> <td>-7.76e-14</td>	Fuel Oil #4 Prod.	5-Methyl chrysene	secondary	-6.99e-13	-7.76e-14
Natural Gas Prod. Benzo[b],k]fluoranthene secondary -8.45e-13 -9.38e-14 Natural Gas Prod. Copper (+1 & +2) secondary -1.06e-12 -1.17e-13 Fuel Oil #4 Prod. Benzo[g,h,j]perylene secondary -1.15e-12 -1.27e-13 Fuel Oil #4 Prod. Benzo[g,h,j]perylene secondary -1.24e-12 -1.37e-13 Fuel Oil #4 Prod. Prod. Pyrene secondary -1.24e-12 -1.37e-13 Fuel Oil #4 Prod. Pyrene secondary -1.24e-12 -1.37e-13 Fuel Oil #4 Prod. Pyrene secondary -1.24e-12 -1.40e-13 Fuel Oil #4 Prod. Fluorene secondary -1.55e-12 -1.70e-13 Natural Gas Prod. Biphenyl secondary -1.55e-12 -1.70e-13 Natural Gas Prod. Chrysene secondary -1.88e-12 -2.02e-13 Natural Gas Prod. Chrysene secondary -1.88e-12 -2.02e-13 Natural Gas Prod. Aromatic hydrocarbons secondary -2.11e-12 -2.34e-13 Natural Gas Pr	Natural Gas Prod.	Pyrene	secondary	-8.10e-13	-8.99e-14
Natural Gas Prod. Copper (+1 & +2) secondary -4.77e-13 -9.72e-14 Natural Gas Prod. Benzo[g,h,i]perylene secondary -1.16e-12 -1.17e-13 Fuel Oil #4 Prod. Benzo[g,h,i]perylene secondary -1.15e-12 -1.27e-13 Fuel Oil #4 Prod. Benzo[g,h,i]perylene secondary -1.24e-12 -1.38e-13 Fuel Oil #4 Prod. Pyrene secondary -1.24e-12 -1.66e-13 Fuel Oil #4 Prod. Benzo[g]/luoranthene secondary -1.24e-12 -1.70e-13 Fuel Oil #4 Prod. Benzo[b]/luoranthene secondary -1.5e-12 -1.72e-13 Fuel Oil #4 Prod. Biphenyl secondary -1.5e-12 -1.72e-13 Natural Gas Prod. Biphenyl secondary -1.64e-12 -1.82e-13 Natural Gas Prod. Chrysene secondary -1.83e-12 -2.02e-13 Natural Gas Prod. Aromatic hydrocarbons secondary -2.32e-12 -2.02e-13 Natural Gas Prod. Acenaphthylene secondary -2.32e-12 -2.76e-13 Natural	Natural Gas Prod.	Benzo[b,j,k]fluoranthene	secondary	-8.45e-13	-9.38e-14
Natural Gas Prod. Benzo[g,h.i]perylene secondary -1.06e-12 -1.17e-13 Fuel Oil #4 Prod. Benzo[g,h.i]perylene secondary -1.15e-12 -1.27e-13 Fuel Oil #4 Prod. Benzo[g,h.i]perylene secondary -1.15e-12 -1.27e-13 Fuel Oil #4 Prod. Pyrene secondary -1.24e-12 -1.38e-13 Fuel Oil #4 Prod. Pyrene secondary -1.25e-12 -1.40e-13 Fuel Oil #4 Prod. Benzo[b]fluoranthene secondary -1.55e-12 -1.70e-13 Stuel Oil #4 Prod. Biphenyl secondary -1.55e-12 -1.72e-13 Fuel Oil #4 Prod. HALON-1301 secondary -1.82e-12 -1.82e-13 Natural Gas Prod. Chrysene secondary -1.87e-12 -2.02e-13 LCD incineration Dibenzo[a,h]anthracene secondary -1.82e-12 -2.02e-13 Natural Gas Prod. Acomatic hydrocarbons secondary -2.11e-12 -2.34e-13 Natural Gas Prod. Acenaphthylene secondary -2.11e-12 -2.37e-13 Natural Gas	Natural Gas Prod.	Copper (+1 & +2)	secondary	-8.77e-13	-9.72e-14
Fuel Oil #4 Prod.Mercury compoundssecondary-1.15e-12-1.27e-13Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.124e-12-1.38e-13Fuel Oil #4 Prod.Pyrenesecondary-1.26e-12-1.40e-13Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.26e-12-1.40e-13Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.53e-12-1.70e-13Natural Gas Prod.Biphenylsecondary-1.55e-12-1.72e-13Fuel Oil #4 Prod.HALON-1301secondary-1.64e-12-1.82e-13Natural Gas Prod.Chrysenesecondary-1.80e-12-1.99e-13LCD incinerationDibenzo[a,h]anthracenesecondary-1.87e-12-2.02e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.11e-12-2.34e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.11e-12-2.34e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.21e-12-2.76e-13Natural Gas Prod.Cumenesecondary-2.61e-12-2.89e-13Natural Gas Prod.Cumenesecondary-2.61e-12-2.89e-13Natural Gas Prod.Chrysenesecondary-2.61e-12-2.89e-13Natural Gas Prod.Chrysenesecondary-2.61e-12-2.89e-13Natural Gas Prod.Chrysenesecondary-2.61e-12-2.89e-13Natural Gas Prod.Benzo[a]anthracenesecondary-2.61e-12-2.89e-13Natural Gas Prod. <td>Natural Gas Prod.</td> <td>Benzo[g,h,i]perylene</td> <td>secondary</td> <td>-1.06e-12</td> <td>-1.17e-13</td>	Natural Gas Prod.	Benzo[g,h,i]perylene	secondary	-1.06e-12	-1.17e-13
Fuel Oil #4 Prod.Benzo[g,h][perylenesecondary-1.15e-12-1.27e-13Fuel Oil #4 Prod.Fluoranthenesecondary-1.26e-12-1.38e-13Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.26e-12-1.40e-13Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.55e-12-1.72e-13Fuel Oil #4 Prod.Fluorenesecondary-1.55e-12-1.72e-13Natural Gas Prod.Biphenylsecondary-1.64e-12-1.82e-13Natural Gas Prod.Chrysenesecondary-1.83e-12-2.02e-13LCD incinerationDibenzo[a,h]anthracenesecondary-1.83e-12-2.02e-13Natural Gas Prod.Aromatic hydrocarbonssecondary-2.18e-12-2.02e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.11e-12-2.34e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.12e-12-2.76e-13Natural Gas Prod.Cumenesecondary-2.21e-12-2.27e-13Natural Gas Prod.Chrysenesecondary-2.61e-12-2.89e-13Fuel Oil #4 Prod.Chrysenesecondary-2.61e-12-2.89e-13Fuel Oil #4 Prod.Chrysenesecondary-2.32e-12-2.27e-13Natural Gas Prod.Chrysenesecondary-2.61e-12-2.89e-13Fuel Oil #4 Prod.Chrysenesecondary-2.61e-12-2.89e-13Fuel Oil #4 Prod.Chrysenesecondary-2.61e-12-2.89e-13Fuel Oil #4 Prod.Benzo[a]a	Fuel Oil #4 Prod.	Mercury compounds	secondary	-1.15e-12	-1.27e-13
Fuel Oil #4 Prod.Fluoranthenesecondary-1.24e-12-1.38e-13Fuel Oil #4 Prod.Pyrenesecondary-1.26e-12-1.40e-13Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.49e-12-1.66e-13Natural Gas Prod.Biphenylsecondary-1.55e-12-1.70e-13Natural Gas Prod.Biphenylsecondary-1.64e-12-1.82e-13Natural Gas Prod.Chrysenesecondary-1.64e-12-1.82e-13Natural Gas Prod.Chrysenesecondary-1.87e-12-2.02e-13Natural Gas Prod.Aromatic hydrocarbonssecondary-1.87e-12-2.02e-13Natural Gas Prod.Aromatic hydrocarbonssecondary-2.11e-12-2.34e-13Natural Gas Prod.Acenaphthylenesecondary-2.32e-12-2.76e-13Natural Gas Prod.Indeno(1,2,3-cd)pyrenesecondary-2.61e-12-2.89e-13Fuel Oil #4 Prod.Chrysenesecondary-2.61e-12-2.89e-13Natural Gas Prod.Cumenesecondary-2.61e-12-2.92e-13Natural Gas Prod.Cumenesecondary-2.61e-12-2.92e-13Natural Gas Prod.Chrysenesecondary-2.61e-12-2.92e-13Natural Gas Prod.Anthracenesecondary-2.61e-12-2.92e-13Natural Gas Prod.Anthracenesecondary-2.61e-12-2.92e-13Natural Gas Prod.Benzo[a]nthracenesecondary-3.03e-12-3.34e-13Natural Gas Prod.Benzo[a]nthracenes	Fuel Oil #4 Prod.	Benzo[g,h,i]perylene	secondary	-1.15e-12	-1.27e-13
Fuel Oil #4 Prod.Pyrenesecondary-1.26e-12-1.40e-13Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.49e-12-1.66e-13Fuel Oil #4 Prod.Fluorenesecondary-1.53e-12-1.70e-13Statural Gas Prod.Biphenylsecondary-1.64e-12-1.82e-13Fuel Oil #4 Prod.HALON-1301secondary-1.64e-12-1.82e-13Natural Gas Prod.Chrysenesecondary-1.83e-12-2.02e-13LCD incinerationDibenzo[a,h]anthracenesecondary-1.87e-12-2.08e-13Natural Gas Prod.Aromatic hydrocarbonssecondary-2.11e-12-2.34e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.32e-12-2.77e-13Natural Gas Prod.Indeno(1,2,3-cd)pyrenesecondary-2.32e-12-2.76e-13Natural Gas Prod.Indeno(1,2,3-cd)pyrenesecondary-2.61e-12-2.89e-13Fuel Oil #4 Prod.Chrysenesecondary-2.61e-12-2.92e-13Natural Gas Prod.Anthracenesecondary-2.61e-12-2.92e-13Natural Gas Prod.Anthracenesecondary-2.63e-12-2.92e-13Natural Gas Prod.Benzo[a]nthracenesecondary-2.38e-12-3.14e-13Fuel Oil #4 Prod.Benzo[a]nthracenesecondary-3.03e-12-3.36e-12Fuel Oil #4 Prod.Benzo[a]nthracenesecondary-3.03e-12-3.42e-13Fuel Oil #4 Prod.Ethyl Chloridesecondary-3.03e-12-3.42e-13Fuel O	Fuel Oil #4 Prod.	Fluoranthene	secondary	-1.24e-12	-1.38e-13
Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.49e-12-1.66e-13Fuel Oil #4 Prod.Fluorenesecondary-1.53e-12-1.70e-13Natural Gas Prod.Biphenylsecondary-1.55e-12-1.72e-13Fuel Oil #4 Prod.HALON-1301secondary-1.64e-12-1.82e-13Natural Gas Prod.Chrysenesecondary-1.83e-12-2.02e-13LCD incinerationDibenzo[a,h]anthracenesecondary-1.87e-12-2.02e-13Natural Gas Prod.Aromatic hydrocarbonssecondary-2.11e-12-2.34e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.11e-12-2.34e-13Natural Gas Prod.Indeno(1,2,3-cd)pyrenesecondary-2.49e-12-2.76e-13Natural Gas Prod.Curnenesecondary-2.63e-12-2.29e-13Natural Gas Prod.Chrysenesecondary-2.63e-12-2.29e-13Natural Gas Prod.Chrysenesecondary-2.63e-12-2.99e-13Natural Gas Prod.Chrysenesecondary-2.63e-12-2.99e-13Sutural Gas Prod.Benzo[a]anthracenesecondary-2.63e-12-2.99e-13Sutural Gas Prod.Benzo[a]anthracenesecondary-2.63e-12-2.99e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.08e-12-3.14e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.08e-12-3.14e-13Fuel Oil #4 Prod.Benzo[b,j,k]fluoranthenesecondary-3.08e-12-3.42e-13<	Fuel Oil #4 Prod.	Pyrene	secondary	-1.26e-12	-1.40e-13
Fuel Oil #4 Prod.Fluorenesecondary-1.53e-12-1.70e-13Natural Gas Prod.Biphenylsecondary-1.55e-12-1.72e-13Fuel Oil #4 Prod.HALON-1301secondary-1.64e-12-1.82e-13Natural Gas Prod.Chrysenesecondary-1.88e-12-2.02e-13Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.87e-12-2.02e-13Natural Gas Prod.Aromatic hydrocarbonssecondary-2.11e-12-2.34e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.32e-12-2.57e-13Natural Gas Prod.Indeno(1,2,3-cd)pyrenesecondary-2.61e-12-2.89e-13Natural Gas Prod.Cumenesecondary-2.61e-12-2.99e-13Natural Gas Prod.Chrysenesecondary-2.69e-12-2.99e-13Natural Gas Prod.Chrysenesecondary-2.69e-12-2.99e-13Natural Gas Prod.Anthracenesecondary-2.69e-12-2.99e-13Natural Gas Prod.Benzo[a]anthracenesecondary-2.69e-12-2.99e-13Natural Gas Prod.Benzo[a]anthracenesecondary-3.08e-12-3.14e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.08e-12-3.14e-13Fuel Oil #4 Prod.Ethyl Chloridesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Benzo[bj,k]fluoranthenesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Benzo[bj,k]fluoranthenesecondary-3.78e-12-4.40e-13F	Fuel Oil #4 Prod.	Benzo[b]fluoranthene	secondary	-1.49e-12	-1.66e-13
Natural Gas Prod.Biphenylsecondary-1.55e-12-1.72e-13Fuel Oil #4 Prod.HALON-1301secondary-1.64e-12-1.82e-13Natural Gas Prod.Chrysenesecondary-1.80e-12-1.99e-13LCD incincrationDibenzo[a,h]anthracenesecondary-1.83e-12-2.02e-13Natural Gas Prod.Aromatic hydrocarbonssecondary-2.11e-12-2.30e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.11e-12-2.34e-13Natural Gas Prod.Acenaphthylenesecondary-2.29e-12-2.57e-13Natural Gas Prod.Indeno(1,2,3-cd)pyrenesecondary-2.61e-12-2.89e-13Vatural Gas Prod.Cumenesecondary-2.61e-12-2.92e-13Natural Gas Prod.Chrysenesecondary-2.61e-12-2.92e-13Natural Gas Prod.Chrysenesecondary-2.61e-12-2.99e-13Vatural Gas Prod.Chrysenesecondary-2.63e-12-2.99e-13Natural Gas Prod.Benzo[a]anthracenesecondary-2.69e-12-2.99e-13Natural Gas Prod.Benzo[a]anthracenesecondary-2.69e-12-2.99e-13Natural Gas Prod.Benzo[a]anthracenesecondary-3.08e-12-3.14e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.08e-12-3.42e-13Natural Gas Prod.Ethyl Chloridesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Benzo[b,jk]fluoranthenesecondary-3.98e-12-3.42e-13 <td< td=""><td>Fuel Oil #4 Prod.</td><td>Fluorene</td><td>secondary</td><td>-1.53e-12</td><td>-1.70e-13</td></td<>	Fuel Oil #4 Prod.	Fluorene	secondary	-1.53e-12	-1.70e-13
Fuel Oil #4 Prod.HALON-1301secondary-1.64e-12-1.82e-13Natural Gas Prod.Chrysenesecondary-1.80e-12-1.99e-13LCD incinerationDibenzo[a,h]anthracenesecondary-1.83e-12-2.02e-13Natural Gas Prod.Aromatic hydrocarbonssecondary-1.87e-12-2.08e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.11e-12-2.34e-13Natural Gas Prod.Acenaphthylenesecondary-2.32e-12-2.57e-13Natural Gas Prod.Indeno(1,2,3-cd)pyrenesecondary-2.61e-12-2.89e-13Natural Gas Prod.Curmenesecondary-2.61e-12-2.92e-13Natural Gas Prod.Chrysenesecondary-2.63e-12-2.92e-13Natural Gas Prod.Chrysenesecondary-2.63e-12-2.92e-13Natural Gas Prod.Chrysenesecondary-2.63e-12-2.99e-13Vatural Gas Prod.Benzo[a]anthracenesecondary-2.69e-12-2.99e-13Natural Gas Prod.Benzo[a]anthracenesecondary-3.03e-12-3.14e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.08e-12-3.42e-13Natural Gas Prod.Ethyl Chloridesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.08e-12-3.42e-13<	Natural Gas Prod.	Biphenyl	secondary	-1.55e-12	-1.72e-13
Natural Gas Prod.Chrysenesecondary-1.80e-12-1.99e-13LCD incinerationDibenzo[a,h]anthracenesecondary-1.83e-12-2.02e-13Natural Gas Prod.Aromatic hydrocarbonssecondary-1.87e-12-2.08e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.11e-12-2.34e-13Natural Gas Prod.Acenaphthylenesecondary-2.32e-12-2.57e-13Natural Gas Prod.Indeno(1,2,3-cd)pyrenesecondary-2.61e-12-2.89e-13Natural Gas Prod.Curmenesecondary-2.61e-12-2.92e-13Natural Gas Prod.Chrysenesecondary-2.63e-12-2.92e-13Natural Gas Prod.Anthracenesecondary-2.69e-12-2.99e-13Natural Gas Prod.Anthracenesecondary-2.69e-12-2.99e-13Natural Gas Prod.Benzo[a]anthracenesecondary-2.83e-12-3.14e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.08e-12-3.42e-13Natural Gas Prod.Indeno(1,2,3-cd)pyrenesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Ethyl Chloridesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Benzo[b_j,k]fluoranthenesecondary-3.92e-12-3.42e-13Fuel Oil #4 Prod.Benzo[b_j,k]fluoranthenesecondary-3.97e-12-3.42e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e	Fuel Oil #4 Prod.	HALON-1301	secondary	-1.64e-12	-1.82e-13
LCD incinerationDibenzo[a,h]anthracenesecondary-1.83e-12-2.02e-13Natural Gas Prod.Aromatic hydrocarbonssecondary-2.11e-12-2.08e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.32e-12-2.57e-13Natural Gas Prod.Indeno(1,2,3-cd)pyrenesecondary-2.49e-12-2.76e-13Natural Gas Prod.Cumenesecondary-2.61e-12-2.89e-13Fuel Oil #4 Prod.Chrysenesecondary-2.63e-12-2.92e-13Natural Gas Prod.Anthracenesecondary-2.69e-12-2.99e-12Natural Gas Prod.Anthracenesecondary-2.69e-12-2.99e-13Natural Gas Prod.Benzo[a]anthracenesecondary-2.83e-12-3.14e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.03e-12-3.03e-12Natural Gas Prod.Benzo[a]anthracenesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Ethyl Chloridesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Benzo[b,j,k]fluoranthenesecondary-3.50e-12-3.88e-13Fuel Oil #4 Prod.Lead cmpdssecondary-3.78e-12-4.19e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.78e-12-4.19e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12 </td <td>Natural Gas Prod.</td> <td>Chrysene</td> <td>secondary</td> <td>-1.80e-12</td> <td>-1.99e-13</td>	Natural Gas Prod.	Chrysene	secondary	-1.80e-12	-1.99e-13
Natural Gas Prod.Aromatic hydrocarbonssecondary-1.87e-12-2.08e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.11e-12-2.34e-13Natural Gas Prod.Acenaphthylenesecondary-2.32e-12-2.57e-13Natural Gas Prod.Indeno(1,2,3-cd)pyrenesecondary-2.49e-12-2.76e-13Natural Gas Prod.Cumenesecondary-2.61e-12-2.89e-13Fuel Oil #4 Prod.Chrysenesecondary-2.63e-12-2.92e-13Natural Gas Prod.Anthracenesecondary-2.63e-12-2.99e-13Natural Gas Prod.Benzo[a]anthracenesecondary-2.83e-12-3.14e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.03e-12-3.36e-13Natural Gas Prod.Ethyl Chloridesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Benzo[b,j,k]fluoranthenesecondary-3.50e-12-3.88e-13Fuel Oil #4 Prod.Benzo[b,j,k]fluoranthenesecondary-3.50e-12-3.88e-13Fuel Oil #4 Prod.Lead cmpdssecondary-3.78e-12-4.19e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary	LCD incineration	Dibenzo[a,h]anthracene	secondary	-1.83e-12	-2.02e-13
Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.11e-12-2.34e-13Natural Gas Prod.Acenaphthylenesecondary-2.32e-12-2.57e-13Natural Gas Prod.Indeno(1,2,3-cd)pyrenesecondary-2.49e-12-2.61e-12Natural Gas Prod.Cumenesecondary-2.61e-12-2.89e-13Fuel Oil #4 Prod.Chrysenesecondary-2.63e-12-2.92e-13Natural Gas Prod.Anthracenesecondary-2.69e-12-2.99e-13Natural Gas Prod.Benzo[a]anthracenesecondary-2.83e-12-3.14e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.03e-12-3.36e-13Natural Gas Prod.Ethyl Chloridesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Nickel cmpdssecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Benzo[b,j,k]fluoranthenesecondary-3.50e-12-3.88e-13Fuel Oil #4 Prod.Benzo[b,j,k]fluoranthenesecondary-3.78e-12-4.19e-13Fuel Oil #4 Prod.Lead cmpdssecondary-3.78e-12-4.40e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-4.32e-12-4.79e-13Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-4.32e-12-4.79e-13Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-	Natural Gas Prod.	Aromatic hydrocarbons	secondary	-1.87e-12	-2.08e-13
Natural Gas Prod.Acenaphthylenesecondary-2.32e-12-2.57e-13Natural Gas Prod.Indeno(1,2,3-cd)pyrenesecondary-2.49e-12-2.76e-13Natural Gas Prod.Cumenesecondary-2.61e-12-2.89e-13Fuel Oil #4 Prod.Chrysenesecondary-2.63e-12-2.92e-13Natural Gas Prod.Anthracenesecondary-2.69e-12-2.99e-13Natural Gas Prod.Benzo[a]anthracenesecondary-2.63e-12-3.14e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.03e-12-3.36e-13Natural Gas Prod.Ethyl Chloridesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Nickel cmpdssecondary-3.50e-12-3.84e-13Fuel Oil #4 Prod.Benzo[b_j,k]fluoranthenesecondary-3.78e-12-3.88e-13Fuel Oil #4 Prod.Lead cmpdssecondary-3.78e-12-4.19e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.78e-12-4.40e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.78e-12-4.40e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-4.32e-12-4.79e-13Natural Gas Prod.1,1,1-Trichloroethanesecondary-4.35e-12-4.82e-13	Natural Gas Prod.	Benzo[b]fluoranthene	secondary	-2.11e-12	-2.34e-13
Natural Gas Prod.Indeno(1,2,3-cd)pyrenesecondary-2.49e-12-2.76e-13Natural Gas Prod.Cumenesecondary-2.61e-12-2.89e-13Fuel Oil #4 Prod.Chrysenesecondary-2.63e-12-2.92e-13Natural Gas Prod.Anthracenesecondary-2.69e-12-2.99e-13Natural Gas Prod.Benzo[a]anthracenesecondary-2.83e-12-3.14e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.03e-12-3.36e-13Natural Gas Prod.Ethyl Chloridesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Nickel cmpdssecondary-3.19e-12-3.54e-13Fuel Oil #4 Prod.Benzo[b,j,k]fluoranthenesecondary-3.50e-12-3.88e-13Fuel Oil #4 Prod.Lead cmpdssecondary-3.78e-12-4.19e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-4.32e-12-4.79e-13Natural Gas Prod.1,1,1-Trichloroethanesecondary-4.35e-12-4.82e-13	Natural Gas Prod.	Acenaphthylene	secondary	-2.32e-12	-2.57e-13
Natural Gas Prod.Cumenesecondary-2.61e-12-2.89e-13Fuel Oil #4 Prod.Chrysenesecondary-2.63e-12-2.92e-13Natural Gas Prod.Anthracenesecondary-2.69e-12-2.99e-13Natural Gas Prod.Benzo[a]anthracenesecondary-2.83e-12-3.14e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.03e-12-3.36e-13Natural Gas Prod.Ethyl Chloridesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Nickel cmpdssecondary-3.19e-12-3.54e-13Fuel Oil #4 Prod.Benzo[b,j,k]fluoranthenesecondary-3.50e-12-3.88e-13Fuel Oil #4 Prod.Lead cmpdssecondary-3.78e-12-4.19e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.78e-12-4.40e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-4.32e-12-4.79e-13Natural Gas Prod.1,1,1-Trichloroethanesecondary-4.32e-12-4.82e-13	Natural Gas Prod.	Indeno(1,2,3-cd)pyrene	secondary	-2.49e-12	-2.76e-13
Fuel Oil #4 Prod.Chrysenesecondary-2.63e-12-2.92e-13Natural Gas Prod.Anthracenesecondary-2.69e-12-2.99e-13Natural Gas Prod.Benzo[a]anthracenesecondary-2.83e-12-3.14e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.03e-12-3.36e-13Natural Gas Prod.Ethyl Chloridesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Nickel cmpdssecondary-3.19e-12-3.54e-13Fuel Oil #4 Prod.Benzo[b,j,k]fluoranthenesecondary-3.50e-12-3.88e-13Fuel Oil #4 Prod.Lead cmpdssecondary-3.78e-12-4.19e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-4.32e-12-4.79e-13Natural Gas Prod.1,1,1-Trichloroethanesecondary-4.35e-12-4.82e-13	Natural Gas Prod.	Cumene	secondary	-2.61e-12	-2.89e-13
Natural Gas Prod.Anthracenesecondary-2.69e-12-2.99e-13Natural Gas Prod.Benzo[a]anthracenesecondary-2.83e-12-3.14e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.03e-12-3.36e-13Natural Gas Prod.Ethyl Chloridesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Nickel cmpdssecondary-3.19e-12-3.54e-13Fuel Oil #4 Prod.Benzo[b,j,k]fluoranthenesecondary-3.50e-12-3.88e-13Fuel Oil #4 Prod.Lead cmpdssecondary-3.78e-12-4.19e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-4.32e-12-4.79e-13Natural Gas Prod.1,1,1-Trichloroethanesecondary-4.35e-12-4.82e-13	Fuel Oil #4 Prod.	Chrysene	secondary	-2.63e-12	-2.92e-13
Natural Gas Prod.Benzo[a]anthracenesecondary-2.83e-12-3.14e-13Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.03e-12-3.36e-13Natural Gas Prod.Ethyl Chloridesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Nickel cmpdssecondary-3.19e-12-3.54e-13Fuel Oil #4 Prod.Benzo[b,j,k]fluoranthenesecondary-3.50e-12-3.88e-13Fuel Oil #4 Prod.Lead cmpdssecondary-3.78e-12-4.19e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-4.32e-12-4.79e-13Natural Gas Prod.1,1,1-Trichloroethanesecondary-4.35e-12-4.82e-13	Natural Gas Prod.	Anthracene	secondary	-2.69e-12	-2.99e-13
Fuel Oil #4 Prod.Indeno(1,2,3-cd)pyrenesecondary-3.03e-12-3.36e-13Natural Gas Prod.Ethyl Chloridesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Nickel cmpdssecondary-3.19e-12-3.54e-13Fuel Oil #4 Prod.Benzo[b,j,k]fluoranthenesecondary-3.50e-12-3.88e-13Fuel Oil #4 Prod.Lead cmpdssecondary-3.78e-12-4.19e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-4.32e-12-4.79e-13Natural Gas Prod.1,1,1-Trichloroethanesecondary-4.35e-12-4.82e-13	Natural Gas Prod.	Benzo[a]anthracene	secondary	-2.83e-12	-3.14e-13
Natural Gas Prod.Ethyl Chloridesecondary-3.08e-12-3.42e-13Fuel Oil #4 Prod.Nickel cmpdssecondary-3.19e-12-3.54e-13Fuel Oil #4 Prod.Benzo[b,j,k]fluoranthenesecondary-3.50e-12-3.88e-13Fuel Oil #4 Prod.Lead cmpdssecondary-3.78e-12-4.19e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-4.32e-12-4.79e-13Natural Gas Prod.1,1,1-Trichloroethanesecondary-4.35e-12-4.82e-13	Fuel Oil #4 Prod.	Indeno(1,2,3-cd)pyrene	secondary	-3.03e-12	-3.36e-13
Fuel Oil #4 Prod.Nickel cmpdssecondary-3.19e-12-3.54e-13Fuel Oil #4 Prod.Benzo[b,j,k]fluoranthenesecondary-3.50e-12-3.88e-13Fuel Oil #4 Prod.Lead cmpdssecondary-3.78e-12-4.19e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-4.32e-12-4.79e-13Natural Gas Prod.1,1,1-Trichloroethanesecondary-4.35e-12-4.82e-13	Natural Gas Prod.	Ethyl Chloride	secondary	-3.08e-12	-3.42e-13
Fuel Oil #4 Prod.Benzo[b,j,k]fluoranthenesecondary-3.50e-12-3.88e-13Fuel Oil #4 Prod.Lead cmpdssecondary-3.78e-12-4.19e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-4.32e-12-4.79e-13Natural Gas Prod.1,1,1-Trichloroethanesecondary-4.35e-12-4.82e-13	Fuel Oil #4 Prod.	Nickel cmpds	secondary	-3.19e-12	-3.54e-13
Fuel Oil #4 Prod.Lead cmpdssecondary-3.78e-12-4.19e-13Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-4.32e-12-4.79e-13Natural Gas Prod.1,1,1-Trichloroethanesecondary-4.35e-12-4.82e-13	Fuel Oil #4 Prod.	Benzo[b,j,k]fluoranthene	secondary	-3.50e-12	-3.88e-13
Fuel Oil #4 Prod.Benzo[a]anthracenesecondary-3.97e-12-4.40e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-4.32e-12-4.79e-13Natural Gas Prod.1,1,1-Trichloroethanesecondary-4.35e-12-4.82e-13	Fuel Oil #4 Prod.	Lead cmpds	secondary	-3.78e-12	-4.19e-13
Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-4.32e-12-4.79e-13Natural Gas Prod.1,1,1-Trichloroethanesecondary-4.35e-12-4.82e-13	Fuel Oil #4 Prod.	Benzo[a]anthracene	secondary	-3.97e-12	-4.40e-13
Natural Gas Prod. 1,1,1-Trichloroethane secondary -4.35e-12 -4.82e-13	Fuel Oil #4 Prod.	Dibenzo[a,h]anthracene	secondary	-4.32e-12	-4.79e-13
	Natural Gas Prod.	1,1,1-Trichloroethane	secondary	-4.35e-12	-4.82e-13

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Cadmium cmpds	secondary	-4.65e-12	-5.16e-13
Fuel Oil #4 Prod.	Anthracene	secondary	-4.72e-12	-5.24e-13
Fuel Oil #4 Prod.	Acenaphthylene	secondary	-4.90e-12	-5.43e-13
Natural Gas Prod.	Acenaphthene	secondary	-5.87e-12	-6.52e-13
Natural Gas Prod.	Dibenzo[a,h]anthracene	secondary	-5.88e-12	-6.52e-13
Fuel Oil #4 Prod.	Biphenyl	secondary	-6.43e-12	-7.13e-13
Natural Gas Prod.	Benzo[a]pyrene	secondary	-6.53e-12	-7.25e-13
Fuel Oil #4 Prod.	Benzo[a]pyrene	secondary	-9.76e-12	-1.08e-12
Natural Gas Prod.	Ethylene dibromide	secondary	-1.06e-11	-1.17e-12
Fuel Oil #4 Prod.	Cumene	secondary	-1.08e-11	-1.20e-12
LCD incineration	2-Methylnaphthalene	secondary	-1.13e-11	-1.25e-12
Natural Gas Prod.	Vinyl acetate	secondary	-1.14e-11	-1.26e-12
Fuel Oil #4 Prod.	Ethyl Chloride	secondary	-1.27e-11	-1.41e-12
LCD incineration	Copper (+1 & +2)	secondary	-1.34e-11	-1.48e-12
Fuel Oil #4 Prod.	Acenaphthene	secondary	-1.51e-11	-1.67e-12
Natural Gas Prod.	Methyl tert-butyl ether	secondary	-1.60e-11	-1.78e-12
Fuel Oil #4 Prod.	1,1,1-Trichloroethane	secondary	-1.80e-11	-1.99e-12
LCD incineration	3-Methylcholanthrene	secondary	-1.80e-11	-2.00e-12
Fuel Oil #4 Prod.	Chlorine	secondary	-2.08e-11	-2.31e-12
Natural Gas Prod.	Phenanthrene	secondary	-2.33e-11	-2.59e-12
Fuel Oil #4 Prod.	2-Methylnaphthalene	secondary	-2.40e-11	-2.66e-12
LCD incineration	Chlorine	secondary	-2.51e-11	-2.79e-12
LCD incineration	Benzo[g,h,i]pervlene	secondary	-2.63e-11	-2.92e-12
Natural Gas Prod.	Aluminum (elemental)	secondary	-2.81e-11	-3.12e-12
LCD incineration	Aromatic hydrocarbons	secondary	-2.86e-11	-3.17e-12
LCD incineration	Zinc (+2)	secondary	-2.90e-11	-3.22e-12
LCD incineration	Dichlorobenzene (mixed isomers)	secondary	-3.18e-11	-3.52e-12
Natural Gas Prod.	Ethylbenzene	secondary	-3.20e-11	-3.55e-12
Natural Gas Prod.	2-Methylnaphthalene	secondary	-3.58e-11	-3.97e-12
Fuel Oil #4 Prod.	3-Methylcholanthrene	secondary	-3.83e-11	-4.25e-12
Fuel Oil #4 Prod.	Zinc (+2)	secondary	-4.24e-11	-4.70e-12
LCD incineration	5-Methyl chrysene	secondary	-4.24e-11	-4.70e-12
Fuel Oil #4 Prod.	Ethylene dibromide	secondary	-4.37e-11	-4.85e-12
Fuel Oil #4 Prod.	Vinvl acetate	secondary	-4.71e-11	-5.23e-12
LCD incineration	Pyrene	secondary	-5.06e-11	-5.62e-12
Fuel Oil #4 Prod.	Phenanthrene	secondary	-5.20e-11	-5.76e-12
Natural Gas Prod.	Chromium (III)	secondary	-5.33e-11	-5.91e-12
Natural Gas Prod.	3-Methylcholanthrene	secondary	-5.72e-11	-6.35e-12
Natural Gas Prod.	Acetophenone	secondary	-5.93e-11	-6.57e-12
Natural Gas Prod	2 4-Dinitrotoluene	secondary	-6 45e-11	-7 16e-12
LCD incineration	Fluoranthene	secondary	-6 52e-11	-7.23e-12
Fuel Oil #4 Prod	Methyl tert-butyl ether	secondary	-6 62e-11	-7 34e-12
Natural Gas Prod	Chlorobenzene	secondary	-8 05e-11	-8.93e-12
LCD incineration	Fluorene	secondary	-8 35e-11	-9.27e-12
LCD incineration	Indeno(1 2 3-cd)nyrene	secondary	-9 04e-11	-1 00e-11
LCD incineration	Chrysene	secondary	-9 73e-11	-1.08e-11
Natural Gas Prod	Dichloromethane	secondary	_9.93e_11	-1 10e-11
Natural Gas Prod	Bromoform	secondary	-1.01e-10	-1 12e-11
Natural Gas Prod	Barium cmnds	secondary	-1.01e-10	-1 18e-11
- (atomi) 000 1 100.	Sarran empas	secondary	1.000-10	1.100 11

 Table M-34.
 LCD LCIA Results for the Chronic Public Health Effects Impact Category

	Material	LCI Data Tuma	Chuania Dublia	/ of Total
Process Group		LCI Data Type	Toxicity (tox-kg)	% 01 10tai
Natural Gas Prod.	Styrene	secondary	-1.08e-10	-1.19e-11
LCD incineration	Benzo[a]anthracene	secondary	-1.18e-10	-1.31e-11
Natural Gas Prod.	Methyl methacrylate	secondary	-1.22e-10	-1.35e-11
Natural Gas Prod.	1,2-Dichloroethane	secondary	-1.25e-10	-1.39e-11
Fuel Oil #4 Prod.	Ethylbenzene	secondary	-1.32e-10	-1.46e-11
Natural Gas Prod.	Tetrachloroethylene	secondary	-1.41e-10	-1.56e-11
Natural Gas Prod.	Methyl ethyl ketone	secondary	-1.43e-10	-1.58e-11
Fuel Oil #4 Prod.	o-xylene	secondary	-1.43e-10	-1.58e-11
Natural Gas Prod.	Isophorone	secondary	-1.78e-10	-1.98e-11
Fuel Oil #4 Prod.	Chromium (III)	secondary	-2.04e-10	-2.27e-11
LCD incineration	Anthracene	secondary	-2.05e-10	-2.28e-11
LCD incineration	Benzo[b,j,k]fluoranthene	secondary	-2.12e-10	-2.35e-11
Fuel Oil #4 Prod.	Copper (+1 & +2)	secondary	-2.12e-10	-2.35e-11
LCD incineration	Benzo[a]pyrene	secondary	-2.31e-10	-2.57e-11
LCD incineration	Acenaphthylene	secondary	-2.41e-10	-2.68e-11
Fuel Oil #4 Prod.	Acetophenone	secondary	-2.45e-10	-2.72e-11
Natural Gas Prod.	Silicon	secondary	-2.46e-10	-2.73e-11
Fuel Oil #4 Prod.	2,4-Dinitrotoluene	secondary	-2.67e-10	-2.96e-11
Natural Gas Prod.	Chlorine	secondary	-3.13e-10	-3.48e-11
Fuel Oil #4 Prod.	Chlorobenzene	secondary	-3.33e-10	-3.69e-11
Natural Gas Prod.	Zinc (+2)	secondary	-3.47e-10	-3.85e-11
Natural Gas Prod.	Dimethyl sulfate	secondary	-3.69e-10	-4.09e-11
LCD incineration	Biphenyl	secondary	-3 90e-10	-4 32e-11
Fuel Oil #4 Prod	Dichloromethane	secondary	-4 11e-10	-4 56e-11
Fuel Oil #4 Prod	Bromoform	secondary	-4 16e-10	-4 62e-11
Fuel Oil #4 Prod	Styrene	secondary	-4 46e-10	-4 94e-11
Fuel Oil #4 Prod	Aromatic hydrocarbons	secondary	-4 53e-10	-5.03e-11
Fuel Oil #4 Prod	Methyl methacrylate	secondary	-5 04e-10	-5 59e-11
Fuel Oil #4 Prod	1 2-Dichloroethane	secondary	-5 19e-10	-5 75e-11
Natural Gas Prod	Acetaldehyde	secondary	-5 30e-10	-5 88e-11
Fuel Oil #4 Prod	Tetrachloroethylene	secondary	-5.83e-10	-6 47e-11
Fuel Oil #4 Prod	Methyl ethyl ketone	secondary	-5 90e-10	-6 54e-11
Natural Gas Prod	Mercury	secondary	-6 23e-10	-6.92e-11
I CD incineration	Cumene	secondary	-6 53e-10	-7 25e-11
LCD incineration	A cenand thene	secondary	-6 59e-10	-7.31e-11
Natural Gas Prod	Di(2-ethylbeyyl)phthalate	secondary	-6.66e-10	-7.39e-11
I CD incineration	Aluminum (elemental)	secondary	-0.00e-10	-7.39e-11
Evel Oil #4 Prod	1 4 Dichlorobenzene	secondary	7 23e 10	-7.79e-11
Fuel Oil #4 Prod	Isonhorone	secondary	7.250-10	-8.02C-11
I CD incineration	Ethyl Chloride	secondary	7.720.10	-8.18e-11
Natural Gas Brad	Mathyl chlorida	secondary	-7.720-10	-8.570-11
Natural Gas Prod	1 4 Dichlorobenzene	secondary	-9.246-10	-1.03e-10
L CD incincration	1,4-Dichlorosthone	secondary	-1.086-09	-1.20e-10
Network Case Due d	A suglein	secondary	-1.096-09	-1.21e-10
Natural Gas Prod.		secondary	-1.11e-09	-1.24e-10
Fuel Oil #4 Prod.	Cadmium empas	secondary	-1.12e-09	-1.25e-10
Fuel OII #4 Prod.		secondary	-1.35e-09	-1.49e-10
	0-xylene	secondary	-1.52e-09	-1.69e-10
Fuel OII #4 Prod.	Dimenyi suitate	secondary	-1.53e-09	-1.69e-10
Natural Gas Prod.		secondary	-1./Se-09	-1.94e-10
Natural Gas Prod.	Aluminum $(+3)$	secondary	-1.90e-09	-2.10e-10
Natural Gas Prod.	Propionaldehyde	secondary	-1.96e-09	-2.18e-10

Fuel Oil #4 Prod.Mercurysecondary-2.05e-09-2.28e-10Natural Gas Prod.Chloroformsecondary-2.12e-09-2.35e-10Fuel Oil #4 Prod.Acetaldehydesecondary-2.19e-09-2.43e-10LCD incinerationXylene (mixed isomers)secondary-2.22e-09-2.46e-10LCD incinerationPhenanthrenesecondary-2.61e-09-2.89e-10LCD incinerationEthylene dibromidesecondary-2.65e-09-2.94e-10Fuel Oil #4 Prod.Di(2-ethylhexyl)phthalatesecondary-2.75e-09-3.05e-10Natural Gas Prod.Toluenesecondary-2.86e-09-3.17e-10LCD incinerationVinyl acetatesecondary-2.86e-09-3.17e-10Natural Gas Prod.Benzyl chloridesecondary-3.27e-09-3.63e-10Fuel Oil #4 Prod.Methyl chloridesecondary-3.27e-09-3.63e-10Natural Gas Prod.Carbon disulfidesecondary-3.93e-09-4.36e-10LCD incinerationMethyl therloridesecondary-3.93e-09-4.36e-10Natural Gas Prod.Carbon disulfidesecondary-3.93e-09-4.36e-10LCD incinerationMethyl tert-butyl ethersecondary-4.61e-09-5.11e-10Natural Gas Prod.Cadmiumsecondary-4.88e-09-5.36e-10LCD incinerationMethyl tert-butyl ethersecondary-5.28e-09-5.36e-10Natural Gas Prod.Cadmiumsecondary-5.28e-09-5.86e-10Natural	Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
Natural Gas Prod.Chloroformsecondary-2.12e-09-2.35e-10Fuel Oil #4 Prod.Acetaldehydesecondary-2.19e-09-2.43e-10LCD incinerationXylene (mixed isomers)secondary-2.22e-09-2.46e-10LCD incinerationPhenanthrenesecondary-2.61e-09-2.89e-10LCD incinerationEthylene dibromidesecondary-2.65e-09-2.94e-10Fuel Oil #4 Prod.Di(2-ethylhexyl)phthalatesecondary-2.75e-09-3.05e-10Natural Gas Prod.Toluenesecondary-2.86e-09-3.17e-10LCD incinerationVinyl acetatesecondary-2.86e-09-3.17e-10Natural Gas Prod.Benzyl chloridesecondary-3.01e-09-3.34e-10Natural Gas Prod.PM-10secondary-3.27e-09-3.63e-10Natural Gas Prod.Carbon disulfidesecondary-3.82e-09-4.24e-10Vatural Gas Prod.Carbon disulfidesecondary-3.93e-09-4.36e-10Natural Gas Prod.Carbon disulfidesecondary-3.93e-09-4.36e-10LCD incinerationMethyl tert-butyl ethersecondary-4.61e-09-5.11e-10Natural Gas Prod.Cadmiumsecondary-5.28e-09-5.36e-10LCD incinerationMethyl tert-butyl ethersecondary-5.28e-09-5.36e-10LCD incinerationMethyl tert-butyl ethersecondary-5.28e-09-5.36e-10Natural Gas Prod.Cadmiumsecondary-5.28e-09-5.36e-10Nat	Fuel Oil #4 Prod.	Mercury	secondary	-2.05e-09	-2.28e-10
Fuel Oil #4 Prod.Acetaldehydesecondary-2.19e-09-2.43e-10LCD incinerationXylene (mixed isomers)secondary-2.22e-09-2.46e-10LCD incinerationPhenanthrenesecondary-2.61e-09-2.89e-10LCD incinerationEthylene dibromidesecondary-2.65e-09-2.94e-10Fuel Oil #4 Prod.Di(2-ethylhexyl)phthalatesecondary-2.75e-09-3.05e-10Natural Gas Prod.Toluenesecondary-2.86e-09-3.17e-10LCD incinerationVinyl acetatesecondary-2.86e-09-3.17e-10Natural Gas Prod.Benzyl chloridesecondary-3.01e-09-3.34e-10Natural Gas Prod.PM-10secondary-3.27e-09-3.63e-10Natural Gas Prod.Carbon disulfidesecondary-3.39e-09-4.26e-10Natural Gas Prod.Carbon disulfidesecondary-3.93e-09-4.26e-10Natural Gas Prod.Carbon disulfidesecondary-3.93e-09-4.36e-10LCD incinerationMethyl tert-butyl ethersecondary-4.61e-09-5.11e-10Natural Gas Prod.Carbon disulfidesecondary-4.61e-09-5.11e-10Natural Gas Prod.Cadmiumsecondary-4.61e-09-5.36e-10LCD incinerationMethyl tert-butyl ethersecondary-4.61e-09-5.36e-10Natural Gas Prod.Cadmiumsecondary-6.14e-09-5.86e-10LCD incinerationSiliconsecondary-6.14e-09-6.81e-10LCD incinerat	Natural Gas Prod.	Chloroform	secondary	-2.12e-09	-2.35e-10
LCD incinerationXylene (mixed isomers)secondary-2.22e-09-2.46e-10LCD incinerationPhenanthrenesecondary-2.61e-09-2.89e-10LCD incinerationEthylene dibromidesecondary-2.65e-09-2.94e-10Fuel Oil #4 Prod.Di(2-ethylhexyl)phthalatesecondary-2.75e-09-3.05e-10Natural Gas Prod.Toluenesecondary-2.86e-09-3.17e-10LCD incinerationVinyl acetatesecondary-2.86e-09-3.17e-10Natural Gas Prod.Benzyl chloridesecondary-3.01e-09-3.34e-10Natural Gas Prod.PM-10secondary-3.27e-09-3.63e-10Fuel Oil #4 Prod.Methyl chloridesecondary-3.82e-09-4.24e-10Natural Gas Prod.Carbon disulfidesecondary-3.93e-09-4.45e-10Fuel Oil #4 Prod.Methyl tert-butyl ethersecondary-3.93e-09-4.45e-10Fuel Oil #4 Prod.Acroleinsecondary-3.93e-09-4.36e-10LCD incinerationMethyl tert-butyl ethersecondary-5.36e-10LCD incinerationMethyl tert-butyl ethersecondary-5.36e-10Natural Gas Prod.Cadmiumsecondary-5.36e-10Natural Gas Prod.Cadmiumsecondary-5.28e-09-5.36e-10LCD incinerationSiliconsecondary-5.28e-09-5.86e-10LCD incinerationSiliconsecondary-6.14e-09-6.81e-10LCD incinerationEthylbenzenesecondary-7.87e-09 </td <td>Fuel Oil #4 Prod.</td> <td>Acetaldehyde</td> <td>secondary</td> <td>-2.19e-09</td> <td>-2.43e-10</td>	Fuel Oil #4 Prod.	Acetaldehyde	secondary	-2.19e-09	-2.43e-10
LCD incinerationPhenanthrenesecondary-2.61e-09-2.89e-10LCD incinerationEthylene dibromidesecondary-2.65e-09-2.94e-10Fuel Oil #4 Prod.Di(2-ethylhexyl)phthalatesecondary-2.75e-09-3.05e-10Natural Gas Prod.Toluenesecondary-2.86e-09-3.17e-10LCD incinerationVinyl acetatesecondary-2.86e-09-3.17e-10Natural Gas Prod.Benzyl chloridesecondary-3.01e-09-3.34e-10Natural Gas Prod.PM-10secondary-3.27e-09-3.63e-10Fuel Oil #4 Prod.Methyl chloridesecondary-3.93e-09-4.24e-10Natural Gas Prod.Carbon disulfidesecondary-3.93e-09-4.36e-10LCD incinerationMethyl tert-butyl ethersecondary-4.01e-09-4.45e-10Fuel Oil #4 Prod.Acroleinsecondary-5.28e-09-5.36e-10LCD incinerationMethyl tert-butyl ethersecondary-5.28e-09-5.36e-10Natural Gas Prod.Cadmiumsecondary-5.28e-09-5.36e-10LCD incinerationSiliconsecondary-6.14e-09-5.86e-10LCD incinerationEthylbenzenesecondary-6.14e-09-6.81e-10LCD incinerationEthylbenzenesecondary-7.87e-09-8.73e-10Fuel Oil #4 Prod.Propionaldehydesecondary-9.00e-10-9.00e-10	LCD incineration	Xylene (mixed isomers)	secondary	-2.22e-09	-2.46e-10
LCD incinerationEthylene dibromidesecondary-2.65e-09-2.94e-10Fuel Oil #4 Prod.Di(2-ethylhexyl)phthalatesecondary-2.75e-09-3.05e-10Natural Gas Prod.Toluenesecondary-2.86e-09-3.17e-10LCD incinerationVinyl acetatesecondary-2.86e-09-3.17e-10Natural Gas Prod.Benzyl chloridesecondary-3.01e-09-3.34e-10Natural Gas Prod.PM-10secondary-3.27e-09-3.63e-10Fuel Oil #4 Prod.Methyl chloridesecondary-3.82e-09-4.24e-10Natural Gas Prod.Carbon disulfidesecondary-3.93e-09-4.36e-10LCD incinerationMethyl tert-butyl ethersecondary-4.61e-09-5.11e-10Natural Gas Prod.Acroleinsecondary-4.61e-09-5.11e-10Natural Gas Prod.Cadmiumsecondary-5.28e-09-5.36e-10LCD incinerationSiliconsecondary-5.28e-09-5.86e-10LCD incinerationSiliconsecondary-6.14e-09-6.81e-10LCD incinerationEthylbenzenesecondary-7.87e-09-8.73e-10Fuel Oil #4 Prod.Propionaldehydesecondary-6.14e-09-6.81e-10	LCD incineration	Phenanthrene	secondary	-2.61e-09	-2.89e-10
Fuel Oil #4 Prod.Di(2-ethylhexyl)phthalatesecondary-2.75e-09-3.05e-10Natural Gas Prod.Toluenesecondary-2.86e-09-3.17e-10LCD incinerationVinyl acetatesecondary-2.86e-09-3.17e-10Natural Gas Prod.Benzyl chloridesecondary-3.01e-09-3.34e-10Natural Gas Prod.PM-10secondary-3.27e-09-3.63e-10Fuel Oil #4 Prod.Methyl chloridesecondary-3.82e-09-4.24e-10Natural Gas Prod.Carbon disulfidesecondary-3.93e-09-4.36e-10LCD incinerationMethyl tert-butyl ethersecondary-4.01e-09-4.45e-10Fuel Oil #4 Prod.Acroleinsecondary-4.61e-09-5.11e-10Natural Gas Prod.Cadmiumsecondary-5.28e-09-5.36e-10LCD incinerationSiliconsecondary-5.28e-09-5.86e-10LCD incinerationSiliconsecondary-6.14e-09-6.81e-10LCD incinerationEthylbenzenesecondary-7.87e-09-8.73e-10Fuel Oil #4 Prod.Prod.Propionaldehydesecondary-6.14e-09-6.81e-10	LCD incineration	Ethylene dibromide	secondary	-2.65e-09	-2.94e-10
Natural Gas Prod.Toluenesecondary-2.86e-09-3.17e-10LCD incinerationVinyl acetatesecondary-2.86e-09-3.17e-10Natural Gas Prod.Benzyl chloridesecondary-3.01e-09-3.34e-10Natural Gas Prod.PM-10secondary-3.27e-09-3.63e-10Fuel Oil #4 Prod.Methyl chloridesecondary-3.82e-09-4.24e-10Natural Gas Prod.Carbon disulfidesecondary-3.93e-09-4.36e-10LCD incinerationMethyl tert-butyl ethersecondary-4.01e-09-4.45e-10Fuel Oil #4 Prod.Acroleinsecondary-4.61e-09-5.11e-10Natural Gas Prod.Cadmiumsecondary-4.61e-09-5.36e-10LCD incinerationMethyl tert-butyl ethersecondary-4.61e-09-5.36e-10Natural Gas Prod.Cadmiumsecondary-5.28e-09-5.86e-10LCD incinerationSiliconsecondary-6.14e-09-6.81e-10LCD incinerationEthylbenzenesecondary-7.87e-09-8.73e-10Fuel Oil #4 Prod.Propionaldehydesecondary-9.00e-10	Fuel Oil #4 Prod.	Di(2-ethylhexyl)phthalate	secondary	-2.75e-09	-3.05e-10
LCD incinerationVinyl acetatesecondary-2.86e-09-3.17e-10Natural Gas Prod.Benzyl chloridesecondary-3.01e-09-3.34e-10Natural Gas Prod.PM-10secondary-3.27e-09-3.63e-10Fuel Oil #4 Prod.Methyl chloridesecondary-3.82e-09-4.24e-10Natural Gas Prod.Carbon disulfidesecondary-3.93e-09-4.36e-10LCD incinerationMethyl tert-butyl ethersecondary-4.01e-09-4.45e-10Fuel Oil #4 Prod.Acroleinsecondary-4.61e-09-5.11e-10Natural Gas Prod.Phenolsecondary-4.83e-09-5.36e-10LCD incinerationSiliconsecondary-6.14e-09-5.86e-10LCD incinerationSiliconsecondary-6.14e-09-6.81e-10LCD incinerationEthylbenzenesecondary-7.87e-09-8.73e-10Fuel Oil #4 Prod.Propionaldehydesecondary-9.00e-10	Natural Gas Prod.	Toluene	secondary	-2.86e-09	-3.17e-10
Natural Gas Prod.Benzyl chloridesecondary-3.01e-09-3.34e-10Natural Gas Prod.PM-10secondary-3.27e-09-3.63e-10Fuel Oil #4 Prod.Methyl chloridesecondary-3.93e-09-4.24e-10Natural Gas Prod.Carbon disulfidesecondary-3.93e-09-4.36e-10LCD incinerationMethyl tert-butyl ethersecondary-4.01e-09-4.45e-10Fuel Oil #4 Prod.Acroleinsecondary-4.61e-09-5.11e-10Natural Gas Prod.Phenolsecondary-4.83e-09-5.36e-10Natural Gas Prod.Cadmiumsecondary-5.28e-09-5.86e-10LCD incinerationSiliconsecondary-6.14e-09-6.81e-10LCD incinerationEthylbenzenesecondary-7.87e-09-8.73e-10Fuel Oil #4 Prod.Propionaldehydesecondary-9.00e-10	LCD incineration	Vinyl acetate	secondary	-2.86e-09	-3.17e-10
Natural Gas Prod.PM-10secondary-3.27e-09-3.63e-10Fuel Oil #4 Prod.Methyl chloridesecondary-3.82e-09-4.24e-10Natural Gas Prod.Carbon disulfidesecondary-3.93e-09-4.36e-10LCD incinerationMethyl tert-butyl ethersecondary-4.01e-09-4.45e-10Fuel Oil #4 Prod.Acroleinsecondary-4.61e-09-5.11e-10Natural Gas Prod.Phenolsecondary-4.83e-09-5.36e-10Natural Gas Prod.Cadmiumsecondary-5.28e-09-5.86e-10LCD incinerationSiliconsecondary-6.14e-09-6.81e-10LCD incinerationEthylbenzenesecondary-7.87e-09-8.73e-10Fuel Oil #4 Prod.Propionaldehydesecondary-9.00e-10	Natural Gas Prod.	Benzyl chloride	secondary	-3.01e-09	-3.34e-10
Fuel Oil #4 Prod.Methyl chloridesecondary-3.82e-09-4.24e-10Natural Gas Prod.Carbon disulfidesecondary-3.93e-09-4.36e-10LCD incinerationMethyl tert-butyl ethersecondary-4.01e-09-4.45e-10Fuel Oil #4 Prod.Acroleinsecondary-4.61e-09-5.11e-10Natural Gas Prod.Phenolsecondary-4.83e-09-5.36e-10Natural Gas Prod.Cadmiumsecondary-5.28e-09-5.86e-10LCD incinerationSiliconsecondary-6.14e-09-6.81e-10LCD incinerationEthylbenzenesecondary-7.87e-09-8.73e-10Fuel Oil #4 Prod.Propionaldehydesecondary-8.11e-09-9.00e-10	Natural Gas Prod.	PM-10	secondary	-3.27e-09	-3.63e-10
Natural Gas Prod.Carbon disulfidesecondary-3.93e-09-4.36e-10LCD incinerationMethyl tert-butyl ethersecondary-4.01e-09-4.45e-10Fuel Oil #4 Prod.Acroleinsecondary-4.61e-09-5.11e-10Natural Gas Prod.Phenolsecondary-4.83e-09-5.36e-10Natural Gas Prod.Cadmiumsecondary-5.28e-09-5.86e-10LCD incinerationSiliconsecondary-6.14e-09-6.81e-10LCD incinerationEthylbenzenesecondary-7.87e-09-8.73e-10Fuel Oil #4 Prod.Propionaldehydesecondary-8.11e-09-9.00e-10	Fuel Oil #4 Prod.	Methyl chloride	secondary	-3.82e-09	-4.24e-10
LCD incinerationMethyl tert-butyl ethersecondary-4.01e-09-4.45e-10Fuel Oil #4 Prod.Acroleinsecondary-4.61e-09-5.11e-10Natural Gas Prod.Phenolsecondary-4.83e-09-5.36e-10Natural Gas Prod.Cadmiumsecondary-5.28e-09-5.86e-10LCD incinerationSiliconsecondary-6.14e-09-6.81e-10LCD incinerationEthylbenzenesecondary-7.87e-09-8.73e-10Fuel Oil #4 Prod.Propionaldehydesecondary-8.11e-09-9.00e-10	Natural Gas Prod.	Carbon disulfide	secondary	-3.93e-09	-4.36e-10
Fuel Oil #4 Prod.Acroleinsecondary-4.61e-09-5.11e-10Natural Gas Prod.Phenolsecondary-4.83e-09-5.36e-10Natural Gas Prod.Cadmiumsecondary-5.28e-09-5.86e-10LCD incinerationSiliconsecondary-6.14e-09-6.81e-10LCD incinerationEthylbenzenesecondary-7.87e-09-8.73e-10Fuel Oil #4 Prod.Propionaldehydesecondary-8.11e-09-9.00e-10	LCD incineration	Methyl tert-butyl ether	secondary	-4.01e-09	-4.45e-10
Natural Gas Prod.Phenolsecondary-4.83e-09-5.36e-10Natural Gas Prod.Cadmiumsecondary-5.28e-09-5.86e-10LCD incinerationSiliconsecondary-6.14e-09-6.81e-10LCD incinerationEthylbenzenesecondary-7.87e-09-8.73e-10Fuel Oil #4 Prod.Propionaldehydesecondary-8.11e-09-9.00e-10	Fuel Oil #4 Prod.	Acrolein	secondary	-4.61e-09	-5.11e-10
Natural Gas Prod.Cadmiumsecondary-5.28e-09-5.86e-10LCD incinerationSiliconsecondary-6.14e-09-6.81e-10LCD incinerationEthylbenzenesecondary-7.87e-09-8.73e-10Fuel Oil #4 Prod.Propionaldehydesecondary-8.11e-09-9.00e-10	Natural Gas Prod.	Phenol	secondary	-4.83e-09	-5.36e-10
LCD incinerationSiliconsecondary-6.14e-09-6.81e-10LCD incinerationEthylbenzenesecondary-7.87e-09-8.73e-10Fuel Oil #4 Prod.Propionaldehydesecondary-8.11e-09-9.00e-10	Natural Gas Prod.	Cadmium	secondary	-5.28e-09	-5.86e-10
LCD incinerationEthylbenzenesecondary-7.87e-09-8.73e-10Fuel Oil #4 Prod.Propionaldehydesecondary-8.11e-09-9.00e-10	LCD incineration	Silicon	secondary	-6.14e-09	-6.81e-10
Fuel Oil #4 Prod.Propionaldehydesecondary-8.11e-09-9.00e-10	LCD incineration	Ethylbenzene	secondary	-7.87e-09	-8.73e-10
	Fuel Oil #4 Prod.	Propionaldehyde	secondary	-8.11e-09	-9.00e-10
Fuel Oil #4 Prod.Chloroformsecondary-8.78e-09-9.74e-10	Fuel Oil #4 Prod.	Chloroform	secondary	-8.78e-09	-9.74e-10
Natural Gas Prod. Cvanide (-1) secondary -9.61e-09 -1.07e-09	Natural Gas Prod.	Cyanide (-1)	secondary	-9.61e-09	-1.07e-09
Natural Gas Prod. Nitrate secondary -9.96e-09 -1.10e-09	Natural Gas Prod.	Nitrate	secondary	-9.96e-09	-1.10e-09
Fuel Oil #4 Prod.Aluminum (elemental)secondary-1.06e-08-1.18e-09	Fuel Oil #4 Prod.	Aluminum (elemental)	secondary	-1.06e-08	-1.18e-09
LCD incineration Chromium (III) secondary -1.13e-08 -1.26e-09	LCD incineration	Chromium (III)	secondary	-1.13e-08	-1.26e-09
Fuel Oil #4 Prod. Cobalt secondary -1.22e-08 -1.36e-09	Fuel Oil #4 Prod.	Cobalt	secondary	-1.22e-08	-1.36e-09
Fuel Oil #4 Prod. Benzyl chloride secondary -1.24e-08 -1.38e-09	Fuel Oil #4 Prod.	Benzyl chloride	secondary	-1.24e-08	-1.38e-09
Natural Gas Prod. Manganese secondary -1.31e-08 -1.45e-09	Natural Gas Prod.	Manganese	secondary	-1.31e-08	-1.45e-09
LCD incineration Chloroacetophenone secondary -1.35e-08 -1.50e-09	LCD incineration	Chloroacetophenone	secondary	-1.35e-08	-1.50e-09
Natural Gas Prod. Beryllium secondary -1.41e-08 -1.57e-09	Natural Gas Prod.	Beryllium	secondary	-1.41e-08	-1.57e-09
Natural Gas Prod. Lead secondary -1.43e-08 -1.59e-09	Natural Gas Prod.	Lead	secondary	-1.43e-08	-1.59e-09
LCD incineration Acetophenone secondary -1.49e-08 -1.65e-09	LCD incineration	Acetophenone	secondary	-1.49e-08	-1.65e-09
LCD incineration 2,4-Dinitrotoluene secondary -1.62e-08 -1.79e-09	LCD incineration	2,4-Dinitrotoluene	secondary	-1.62e-08	-1.79e-09
Fuel Oil #4 Prod. Carbon disulfide secondary -1.63e-08 -1.80e-09	Fuel Oil #4 Prod.	Carbon disulfide	secondary	-1.63e-08	-1.80e-09
Natural Gas Prod. Copper secondary -1.75e-08 -1.94e-09	Natural Gas Prod.	Copper	secondary	-1.75e-08	-1.94e-09
Natural Gas Prod. Bromomethane secondary -1.83e-08 -2.03e-09	Natural Gas Prod.	Bromomethane	secondary	-1.83e-08	-2.03e-09
Natural Gas Prod. 2-Chloroacetophenone secondary -1.85e-08 -2.05e-09	Natural Gas Prod.	2-Chloroacetophenone	secondary	-1.85e-08	-2.05e-09
Natural Gas Prod. Barium secondary -1.95e-08 -2.17e-09	Natural Gas Prod.	Barium	secondary	-1.95e-08	-2.17e-09
Natural Gas Prod. Methyl hydrazine secondary -1.98e-08 -2.19e-09	Natural Gas Prod.	Methyl hydrazine	secondary	-1.98e-08	-2.19e-09
LCD incineration Chlorobenzene secondary -2.02e-08 -2.24e-09	LCD incineration	Chlorobenzene	secondary	-2.02e-08	-2.24e-09
Fuel Oil #4 Prod. Cadmium secondary -2.16e-08 -2.39e-09	Fuel Oil #4 Prod.	Cadmium	secondary	-2.16e-08	-2.39e-09
LCD incineration Dichloromethane secondary -2.49e-08 -2.76e-09	LCD incineration	Dichloromethane	secondary	-2.49e-08	-2.76e-09
LCD incineration Bromoform secondary -2.52e-08 -2.80e-09	LCD incineration	Bromoform	secondary	-2.52e-08	-2.80e-09
Fuel Oil #4 Prod. Barium cmpds secondary -2.57e-08 -2.85e-09	Fuel Oil #4 Prod.	Barium cmpds	secondary	-2.57e-08	-2.85e-09
LCD incineration Styrene secondary -2.70e-08 -3.00e-09	LCD incineration	Styrene	secondary	-2.70e-08	-3.00e-09
LCD incineration Aluminum (+3) secondary -2.89e-08 -3.20e-09	LCD incineration	Aluminum (+3)	secondary	-2.89e-08	-3.20e-09
LCD incineration Methyl methacrylate secondary -3.06e-08 -3.39e-09	LCD incineration	Methyl methacrylate	secondary	-3.06e-08	-3.39e-09
LCD incineration 1,2-Dichloroethane secondary -3.12e-08 -3.46e-09	LCD incineration	1,2-Dichloroethane	secondary	-3.12e-08	-3.46e-09
LCD incineration Dimethylbenzanthracene secondary -3.12e-08 -3.47e-09	LCD incineration	Dimethylbenzanthracene	secondary	-3.12e-08	-3.47e-09

 Table M-34.
 LCD LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
LCD incineration	Tetrachloroethylene	secondary	-3.51e-08	-3.90e-09
LCD incineration	Methyl ethyl ketone	secondary	-3.58e-08	-3.97e-09
Fuel Oil #4 Prod.	Barium	secondary	-3.61e-08	-4.00e-09
Fuel Oil #4 Prod.	Cyanide (-1)	secondary	-3.97e-08	-4.41e-09
Natural Gas Prod.	Naphthalene	secondary	-4.37e-08	-4.84e-09
Natural Gas Prod.	Hydrogen sulfide	secondary	-4.46e-08	-4.94e-09
LCD incineration	Isophorone	secondary	-4.47e-08	-4.96e-09
LCD incineration	Toluene	secondary	-4.73e-08	-5.24e-09
Fuel Oil #4 Prod.	Beryllium	secondary	-5.06e-08	-5.62e-09
Fuel Oil #4 Prod.	Manganese	secondary	-5.28e-08	-5.86e-09
Fuel Oil #4 Prod.	Naphthalene	secondary	-5.36e-08	-5.94e-09
Natural Gas Prod.	Nickel	secondary	-5.78e-08	-6.41e-09
Fuel Oil #4 Prod.	Lead	secondary	-5.88e-08	-6.52e-09
Fuel Oil #4 Prod.	Dimethylbenzanthracene	secondary	-6.37e-08	-7.06e-09
Fuel Oil #4 Prod.	Bromomethane	secondary	-7.56e-08	-8.39e-09
LCD incineration	Phenol	secondary	-7.60e-08	-8.43e-09
Fuel Oil #4 Prod.	2-Chloroacetophenone	secondary	-7.65e-08	-8.49e-09
LCD incineration	PM-10	secondary	-8.16e-08	-9.05e-09
Fuel Oil #4 Prod.	Methyl hydrazine	secondary	-8.17e-08	-9.06e-09
LCD incineration	Mercury	secondary	-8.42e-08	-9.34e-09
Fuel Oil #4 Prod.	Copper	secondary	-8.86e-08	-9.82e-09
LCD incineration	Dimethyl sulfate	secondary	-9.25e-08	-1.03e-08
Fuel Oil #4 Prod.	Silicon	secondary	-9.29e-08	-1.03e-08
Natural Gas Prod.	Dimethylbenzanthracene	secondary	-9.51e-08	-1.06e-08
LCD incineration	Acetaldehvde	secondary	-1.33e-07	-1.47e-08
Natural Gas Prod.	Antimony	secondary	-1.48e-07	-1.64e-08
LCD incineration	Di(2-ethylhexyl)phthalate	secondary	-1.67e-07	-1.85e-08
Fuel Oil #4 Prod.	Nitrate	secondary	-2.08e-07	-2.31e-08
LCD incineration	Methyl chloride	secondary	-2.32e-07	-2.57e-08
LCD incineration	Cobalt	secondary	-2.48e-07	-2.76e-08
LCD incineration	Copper	secondary	-2.67e-07	-2.96e-08
LCD incineration	Acrolein	secondary	-2.79e-07	-3.10e-08
Fuel Oil #4 Prod.	Antimony	secondary	-4.22e-07	-4.68e-08
Fuel Oil #4 Prod.	Aluminum (+3)	secondary	-4.59e-07	-5.09e-08
Natural Gas Prod.	Chromium (VI)	secondary	-4.90e-07	-5.43e-08
LCD incineration	Propionaldehyde	secondary	-4.92e-07	-5.46e-08
LCD incineration	Chloroform	secondary	-5.30e-07	-5.88e-08
Natural Gas Prod	Phosphorus (vellow or white)	secondary	-5 86e-07	-6 50e-08
LCD incineration	Pentane	secondary	-6.12e-07	-6.78e-08
LCD incineration	Cadmium	secondary	-7 25e-07	-8 04e-08
LCD incineration	Benzyl chloride	secondary	-7 55e-07	-8 37e-08
Natural Gas Prod	Molybdenum	secondary	-7 66e-07	-8 50e-08
LCD incineration	Barium	secondary	-8 87e-07	-9 84e-08
LCD incineration	Hexane	secondary	-9 76e-07	-1 08e-07
I CD incineration	Carbon disulfide	secondary	-9.86e-07	-1.09e-07
LCD incineration	Hydrogen sulfide	secondary	-1 03e-06	-1 14e-07
Fuel Oil #4 Prod	Nickel	secondary	-1.09e-06	-1 20e-07
Natural Gas Prod	Hydrofluoric acid	secondary	-1.15e.06	-1.280-07
Fuel Oil #4 Prod	Phenol	secondary	-1.150-00	-1.28C-07
Fuel Oil #4 Prod	PM-10	secondary	-1.100-00	-1.376-07
Fuel Oil #4 Prod	Pentane	secondary	-1.240-00	-1.576-07
ruci ()11 #4 r100.	I UIRANG	secondary	-1.508-00	-1.446-07

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
LCD incineration	Naphthalene	secondary	-1.33e-06	-1.48e-07
LCD incineration	Ethane	secondary	-1.46e-06	-1.62e-07
Fuel Oil #4 Prod.	Hexane	secondary	-1.80e-06	-2.00e-07
Fuel Oil #4 Prod.	Chromium (VI)	secondary	-1.88e-06	-2.08e-07
Natural Gas Prod.	Pentane	secondary	-1.94e-06	-2.15e-07
LCD incineration	Nitrate	secondary	-2.33e-06	-2.58e-07
Fuel Oil #4 Prod.	Zinc (elemental)	secondary	-2.35e-06	-2.61e-07
LCD incineration	Cyanide (-1)	secondary	-2.41e-06	-2.67e-07
Natural Gas Prod.	Hexane	secondary	-2.69e-06	-2.98e-07
Natural Gas Prod.	Zinc (elemental)	secondary	-2.81e-06	-3.12e-07
LCD incineration	Beryllium	secondary	-2.97e-06	-3.30e-07
Fuel Oil #4 Prod.	Ethane	secondary	-3.10e-06	-3.43e-07
LCD incineration	Manganese	secondary	-3.17e-06	-3.52e-07
Natural Gas Prod.	Selenium	secondary	-4.02e-06	-4.46e-07
Fuel Oil #4 Prod.	Molybdenum	secondary	-4.28e-06	-4.75e-07
Natural Gas Prod.	Fluorides (F-)	secondary	-4.50e-06	-4.99e-07
LCD incineration	Bromomethane	secondary	-4.59e-06	-5.09e-07
Natural Gas Prod.	Ethane	secondary	-4.63e-06	-5.13e-07
Fuel Oil #4 Prod	Hydrofluoric acid	secondary	-4 77e-06	-5 29e-07
LCD incineration	Methyl hydrazine	secondary	-4 95e-06	-5 49e-07
LCD incineration	Molybdenum	secondary	-6 45e-06	-7 16e-07
Natural Gas Prod	Nitrous oxide	secondary	-7 65e-06	-8 48e-07
I CD incineration	Nickel	secondary	-8 94e-06	-9.91e-07
Natural Gas Prod	Formaldehyde	secondary	-9 17e-06	-1.02e-06
Fuel Oil #4 Prod	Ammonia	secondary	-1 15e-05	-1 27e-06
I CD incineration	Antimony	secondary	-1 40e-05	-1 55e-06
LCD incineration	Zinc (elemental)	secondary	-1.65e-05	-1 83e-06
Fuel Oil #4 Prod	Hydrogen sulfide	secondary	-1.68e-05	-1 87e-06
Fuel Oil #4 Prod	Fluorides (F-)	secondary	-1.76e-05	-1.95e-06
Fuel Oil #4 Prod	Selenium	secondary	-1.70c-05	-1.97e-06
Natural Gas Prod	Hydrochloric acid	secondary	-2 11e-05	-2 35e-06
Natural Gas Prod	Vanadium	secondary	-2 30e-05	-2.55e 00
Fuel Oil #4 Prod	Phosphorus (vellow or white)	secondary	-2.500-05	-2.90e-00
I CD incineration	Phosphorus (yellow or white)	secondary	-3 56e-05	-3.95e-06
LCD incineration	Ammonia	secondary	-5.500-05	5 340 06
Fuel Oil #4 Prod	Nitrous oxide	secondary	-4.820-05	-9.05e-06
Fuel Oil #4 Prod	Hydrochloric acid	secondary	-8.10c-05	-9.69e-06
I CD incineration	Chromium (VI)	secondary	-8.740-03	1 15e 05
Natural Gas Prod	Arsenic	secondary	-1.04c-04	-1.15c-05
Natural Gas Prod	Ammonia	secondary	1 /10 04	1 56e 05
Natural Gas Prod	Sulfur oxides	secondary	-1.410-04	-1.500-05
Natural Cas Prod	DM	secondary	-1.050-04	-2.020-05
LCD incincration	r w	secondary	-2.746-04	-3.046-03
LCD incincration	Formaldahyda	secondary	-2.696-04	-3.210-05
Evol Oil #4 Prod	Formaldehyde	secondary	-2.908-04	-5.210-05
Fuel Oil #4 Prod	Amonio	secondary	-3.03e-04	-4.03e-05
Fuel Oil #4 Prod	DM	secondary	-3.008-04	-5.550-05
L CD incincention	r IVI Vone dive	secondary	-5./5e-04	-0.5/e-05
LCD incineration	vanaulum Selenium	secondary	-/.60e-04	-8.43e-05
LCD incineration	Selenium	secondary	-9.94e-04	-1.10e-04

 Table M-34.
 LCD LCIA Results for the Chronic Public Health Effects Impact Category

Process Group	Material	LCI Data Type	Chronic Public Toxicity (tox-kg)	% of Total
LCD incineration	Fluorides (F-)	secondary	-1.02e-03	-1.13e-04
LCD incineration	Nitrous oxide	secondary	-1.35e-03	-1.50e-04
Fuel Oil #4 Prod.	Benzene	secondary	-1.71e-03	-1.90e-04
LCD incineration	Benzene	secondary	-2.08e-03	-2.31e-04
Fuel Oil #4 Prod.	Nitrogen oxides	secondary	-2.55e-03	-2.83e-04
Fuel Oil #4 Prod.	Vanadium	secondary	-2.86e-03	-3.17e-04
LCD incineration	Hydrochloric acid	secondary	-3.08e-03	-3.41e-04
Fuel Oil #4 Prod.	Sulfur oxides	secondary	-3.48e-03	-3.86e-04
Fuel Oil #4 Prod.	Methane	secondary	-3.70e-03	-4.10e-04
Natural Gas Prod.	Nitrogen oxides	secondary	-6.17e-03	-6.85e-04
Fuel Oil #4 Prod.	Carbon monoxide	secondary	-1.08e-02	-1.19e-03
LCD incineration	Carbon monoxide	secondary	-1.55e-02	-1.72e-03
LCD incineration	Nitrogen oxides	secondary	-1.86e-02	-2.07e-03
LCD incineration	Methane	secondary	-1.97e-02	-2.18e-03
Natural Gas Prod.	Carbon monoxide	secondary	-2.27e-02	-2.52e-03
LCD incineration	PM	secondary	-2.40e-02	-2.66e-03
LCD incineration	Arsenic	secondary	-2.50e-02	-2.78e-03
Natural Gas Prod.	Benzene	secondary	-2.51e-02	-2.78e-03
Natural Gas Prod.	Methane	secondary	-3.32e-02	-3.68e-03
LCD incineration	Sulfur oxides	secondary	-3.50e-02	-3.88e-03
Total End-of-life			-6.30e-02	-6.99e-03
Total All Life-cycle Stages			9.02e+02	1.00e+02

	CITA Results for the rest	ierres impacts c	ategory	
Process Group	Material	LCI Data Type	Odor (m3)	% of Total
Materials Processing Life-cycle Stage				
Steel Prod., cold-rolled, semi-finished	Hydrogen sulfide	secondary	3.17e+04	4.18e-01
ABS Production	Hydrogen sulfide	secondary	2.36e+04	3.11e-01
Invar	Hydrogen sulfide	secondary	1.69e+04	2.24e-01
Ferrite mfg.	Hydrogen sulfide	secondary	1.21e+04	1.59e-01
Polycarbonate Production	Ethanethiol	secondary	1.05e+04	1.39e-01
Aluminum Prod.	Hydrogen sulfide	secondary	8.78e+03	1.16e-01
Lead	Hydrogen sulfide	secondary	6.91e+03	9.12e-02
ABS Production	Ethanethiol	secondary	4.82e+03	6.37e-02
Polycarbonate Production	Hydrogen sulfide	secondary	4.30e+03	5.68e-02
Invar	Acetaldehyde	secondary	4.27e+03	5.64e-02
Steel Prod., cold-rolled, semi-finished	Acetaldehyde	secondary	2.48e+03	3.28e-02
Ferrite mfg.	Acetaldehyde	secondary	7.11e+02	9.38e-03
Lead	Acetaldehyde	secondary	5.50e+02	7.26e-03
Styrene-butadiene Copolymer Prod.	Hydrogen sulfide	secondary	4.82e+02	6.37e-03
Steel Prod., cold-rolled, semi-finished	Ammonia	secondary	4.15e+02	5.48e-03
Aluminum Prod.	Acetic acid	secondary	1.14e+02	1.50e-03
Invar	Acetic acid	secondary	9 41e+01	1 24e-03
Steel Prod., cold-rolled, semi-finished	Acetic acid	secondary	5.51e+01	7.27e-04
Steel Prod cold-rolled semi-finished	Phenol	secondary	4 82e+01	6 37e-04
Aluminum Prod	Xylene (mixed isomers)	secondary	3 04e+01	4 01e-04
ABS Production	Formaldehyde	secondary	2 86e+01	3 77e-04
Ferrite mfg	Acetic acid	secondary	2.35e+01	3.10e-04
Lead	Acetic acid	secondary	1.96e+01	2 59e-04
Aluminum Prod	Toluene	secondary	1.39e+01	1.83e-04
Invar	Phenol	secondary	1.37e+01	1.83e-04
Ferrite mfg	Phenol	secondary	1.37e+01	1.01e 01
Steel Prod cold-rolled semi-finished	Toluene	secondary	1 30e+01	1.70 c 01
Invar	Xylene (mixed isomers)	secondary	1.19e+01	1.58e-04
Aluminum Prod	Ammonia	secondary	9.03e+00	1.50e 04
Steel Prod. cold-rolled semi-finished	Xylene (mixed isomers)	secondary	8.98e+00	1.19e-04
ABS Production	Fthanol	secondary	8.17e+00	1.08e-04
Steel Prod. cold-rolled semi-finished	Formaldehyde	secondary	7.41e+00	9.78e-05
Lead	Ammonia	secondary	6.93e+00	9.15e-05
Lead	Xylene (mixed isomers)	secondary	5.59e+00	7.37e-05
Invor	Toluene	secondary	4.74e+00	6 250 05
Ferrite mfg	Toluene	secondary	4.740+00	6 10e 05
Invor	Ethanol	secondary	3 58e+00	4 73e 05
Invar	Ammonia	secondary	2.55c+00	4.730-05
Formite mfg	Annonia Vulene (mived isomers)	secondary	2 100+00	4.096-05
Ferrier Inig.	Earmaldabyda	secondary	3.10e+00	2.770.05
	Formaldenyde	secondary	2.800+00	3.776-03
Ferrite mig.	Formaldenyde	secondary	2.79e+00	3.086-05
Steel Prod., cold-rolled, semi-linished		secondary	2.08e+00	2.74e-05
Polycarbonate Production	Chlorine	secondary	2.01e+00	2.66e-05
Invar	Ethanethiol	secondary	1.98e+00	2.62e-05
Ferrite mfg.	Ethanethiol	secondary	1.94e+00	2.56e-05
ABS Production	Chlorine	secondary	1.85e+00	2.44e-05
Ferrite mig.	Ammonia	secondary	1.17e+00	1.55e-05
Styrene-butadiene Copolymer Prod.	Chlorine	secondary	9.01e-01	1.19e-05
ABS Production	Ammonia	secondary	8.49e-01	1.12e-05
Invar	Methanol	secondary	6.43e-01	8.49e-06

Process Group	Material	LCI Data Type	Odor (m3)	% of Total
Ferrite mfg.	Ethanol	secondary	5.85e-01	7.73e-06
Lead	Ethanol	secondary	4.54e-01	5.99e-06
Steel Prod., cold-rolled, semi-finished	Methanol	secondary	4.11e-01	5.43e-06
Steel Prod., cold-rolled, semi-finished	Benzene	secondary	2.58e-01	3.41e-06
Styrene-butadiene Copolymer Prod.	Ammonia	secondary	2.07e-01	2.74e-06
Invar	Benzene	secondary	1.27e-01	1.68e-06
Ferrite mfg.	Methanol	secondary	1.16e-01	1.53e-06
Aluminum Prod.	Phenol	secondary	1.03e-01	1.36e-06
Ferrite mfg.	Benzene	secondary	8.64e-02	1.14e-06
Steel Prod., cold-rolled, semi-finished	Naphthalene	secondary	7.46e-02	9.84e-07
Aluminum Prod.	Benzene	secondary	6.00e-02	7.92e-07
Lead	Methanol	secondary	5.60e-02	7.39e-07
Lead	Benzene	secondary	4.09e-02	5.40e-07
Invar	Propionic acid	secondary	3.92e-02	5.18e-07
Invar	Acetone	secondary	1.59e-02	2.10e-07
Steel Prod., cold-rolled, semi-finished	Acetone	secondary	9.25e-03	1.22e-07
Ferrite mfg.	Acetone	secondary	2.61e-03	3.44e-08
Lead	Acetone	secondary	2.01e-03	2.65e-08
Invar	Chlorine	secondary	6.89e-04	9.09e-09
Ferrite mfg.	Chlorine	secondary	6.72e-04	8.88e-09
Steel Prod., cold-rolled, semi-finished	Isopropylpropionate	secondary	6.34e-04	8.36e-09
Ferrite mfg.	Isopropylpropionate	secondary	6.22e-04	8.21e-09
Lead	Acrolein	secondary	3.46e-04	4.57e-09
Steel Prod., cold-rolled, semi-finished	Chlorine	secondary	8.13e-05	1.07e-09
Invar	Propionaldehyde	secondary	4.43e-05	5.85e-10
Steel Prod., cold-rolled, semi-finished	Propionaldehyde	secondary	4.43e-05	5.85e-10
Ferrite mfg.	Propionaldehyde	secondary	4.33e-05	5.71e-10
Invar	Acrolein	secondary	1.44e-07	1.90e-12
Total Materials Processing			1.29e+05	1.70e+00
Manufacturing Life-cycle Stage				
LPG Production	Hydrogen sulfide	secondary	7.15e+06	9.43e+01
Fuel Oil #6 Prod.	Hydrogen sulfide	secondary	7.40e+04	9.77e-01
Fuel Oil #2 Prod.	Hydrogen sulfide	secondary	2.36e+04	3.11e-01
LPG Production	Acetaldehyde	secondary	1.42e+04	1.87e-01
CRT tube mfg.	Toluene	primary	6.41e+03	8.46e-02
US electric grid	Acetaldehyde	model/secondary	3.03e+03	4.00e-02
Fuel Oil #4 Prod.	Hydrogen sulfide	secondary	2.76e+03	3.65e-02
Japanese Electric Grid	Acetaldehyde	model/secondary	2.42e+03	3.19e-02
LPG Production	Formaldehyde	secondary	2.39e+03	3.16e-02
LPG Production	Ammonia	secondary	2.18e+03	2.88e-02
LPG Production	Propionaldehyde	secondary	7.28e+02	9.61e-03
CRT tube mfg.	Xylene (mixed isomers)	primary	6.31e+02	8.33e-03
Natural Gas Prod.	Ammonia	secondary	1.57e+02	2.07e-03
Natural Gas Prod.	Hydrogen sulfide	secondary	1.57e+02	2.07e-03
US electric grid	Propionaldehyde	model/secondary	1.56e+02	2.06e-03
LPG Production	Benzene	secondary	1.33e+02	1.76e-03
Japanese Electric Grid	Propionaldehyde	model/secondary	1.24e+02	1.64e-03
Fuel Oil #6 Prod.	Acetaldehyde	secondary	1.14e+02	1.51e-03
PWB Mfg.	Formaldehyde	model/secondary	7.94e+01	1.05e-03
Fuel Oil #2 Prod.	Acetaldehyde	secondary	4.44e+01	5.86e-04
LPG Production	Acrolein	secondary	2.82e+01	3.72e-04

Process Group	Material	LCI Data Type	Odor (m3)	% of Total
Natural Gas Prod.	Acetaldehyde	secondary	2.46e+01	3.24e-04
Fuel Oil #6 Prod.	Formaldehyde	secondary	2.45e+01	3.23e-04
LPG Production	Benzyl chloride	secondary	2.24e+01	2.95e-04
Japanese Electric Grid	Formaldehyde	model/secondary	2.09e+01	2.76e-04
Natural Gas Prod.	Benzene	secondary	1.18e+01	1.55e-04
Fuel Oil #6 Prod.	Ammonia	secondary	1.17e+01	1.55e-04
Fuel Oil #2 Prod.	Formaldehyde	secondary	7.87e+00	1.04e-04
Fuel Oil #2 Prod.	Ammonia	secondary	6.43e+00	8.49e-05
LPG Production	Toluene	secondary	6.21e+00	8.19e-05
US electric grid	Acrolein	model/secondary	6.03e+00	7.96e-05
Fuel Oil #6 Prod.	Propionaldehyde	secondary	5.87e+00	7.76e-05
LPG Production	Carbon disulfide	secondary	4.84e+00	6.39e-05
Japanese Electric Grid	Acrolein	model/secondary	4.81e+00	6.36e-05
US electric grid	Benzyl chloride	model/secondary	4.78e+00	6.31e-05
Fuel Oil #4 Prod.	Acetaldehyde	secondary	4.74e+00	6.25e-05
LPG Production	Methyl ethyl ketone	secondary	3.85e+00	5.08e-05
Japanese Electric Grid	Benzyl chloride	model/secondary	3.82e+00	5.04e-05
LPG Production	Isophorone	secondary	3.54e+00	4.67e-05
US electric grid	Formaldehvde	model/secondary	2.76e+00	3.65e-05
LPG Production	Phenol	secondary	2.75e+00	3.63e-05
LPG Production	Styrene	secondary	2.47e+00	3.25e-05
Japanese Electric Grid	Toluene	model/secondary	2.39e+00	3.15e-05
Fuel Oil #2 Prod.	Propionaldehvde	secondary	2.28e+00	3.02e-05
LPG Production	Naphthalene	secondary	1.75e+00	2.31e-05
LPG Production	o-xylene	secondary	1.33e+00	1.75e-05
Natural Gas Prod.	Propionaldehvde	secondary	1.26e+00	1.67e-05
Japanese Electric Grid	Naphthalene	model/secondary	1.11e+00	1.46e-05
US electric grid	Carbon disulfide	model/secondary	1.04e+00	1.37e-05
Fuel Oil #4 Prod	Formaldehyde	secondary	9 18e-01	1 21e-05
LPG Production	Cumene	secondary	8 89e-01	1 17e-05
Japanese Electric Grid	Carbon disulfide	model/secondary	8.27e-01	1.09e-05
US electric grid	Methyl ethyl ketone	model/secondary	8 23e-01	1.09e-05
Fuel Oil #6 Prod	Benzene	secondary	7.61e-01	1.00e-05
US electric orid	Isophorone	model/secondary	7 57e-01	9.99e-06
US electric grid	Toluene	model/secondary	6 78e-01	8 95e-06
LPG Production	Methyl methacrylate	secondary	6 71e-01	8 85e-06
Jananese Electric Grid	Methyl ethyl ketone	model/secondary	6 57e-01	8.67e-06
Japanese Electric Grid	Isophorone	model/secondary	6.04e-01	7.97e-06
LPG Production	Methyl hydrazine	secondary	6 00e-01	7.92e-06
Fuel Oil #4 Prod	Ammonia	secondary	5.97e-01	7.88e-06
US electric grid	Phenol	model/secondary	5.89e-01	7.00e 00
US electric grid	Styrene	model/secondary	5.05e 01	6.96e-06
Natural Gas Prod	Formaldehyde	secondary	4 95e-01	6.53e-06
Jananese Electric Grid	Phenol	model/secondary	4.70e-01	6.20e-06
L PG Production	2-Chloroacetonhenone	secondary	4.69e-01	6.20e-06
Jananese Electric Grid	Styrana	model/secondary	4.050-01	5 569 06
I PG Production	1 4-Dichlorobenzene	secondary	4 120-01	5.30C-00
Fuel Oil #2 Prod	Renzene	secondary	3.96e-01	5.77e-06
Fuel Oil #4 Prod	Propionaldehyde	secondary	2.440.01	3 220 04
Fuel Oil #6 Prod	A crolein	secondary	2.440-01	3.000.06
LUS alastria grid	Cumono	model/seeenderry	1.002.01	2.51 - 00
US electric grid	Cumene	model/secondary	1.90e-01	2.51e-06

Process Group	Material	LCI Data Type	Odor (m3)	% of Total
Fuel Oil #6 Prod.	Benzyl chloride	secondary	1.80e-01	2.38e-06
LPG Production	Chlorobenzene	secondary	1.48e-01	1.95e-06
US electric grid	Naphthalene	model/secondary	1.46e-01	1.93e-06
US electric grid	Methyl methacrylate	model/secondary	1.43e-01	1.89e-06
US electric grid	Methyl hydrazine	model/secondary	1.28e-01	1.69e-06
LPG Production	Vinyl acetate	secondary	1.27e-01	1.68e-06
Japanese Electric Grid	Methyl methacrylate	model/secondary	1.15e-01	1.51e-06
Japanese Electric Grid	Methyl hydrazine	model/secondary	1.03e-01	1.35e-06
US electric grid	2-Chloroacetophenone	model/secondary	1.00e-01	1.33e-06
Natural Gas Prod.	o-xylene	secondary	8.89e-02	1.17e-06
Fuel Oil #2 Prod.	Acrolein	secondary	8.84e-02	1.17e-06
US electric grid	Xylene (mixed isomers)	model/secondary	8.70e-02	1.15e-06
Natural Gas Prod.	Toluene	secondary	8.63e-02	1.14e-06
Japanese Electric Grid	2-Chloroacetophenone	model/secondary	8.02e-02	1.06e-06
Fuel Oil #2 Prod.	Benzyl chloride	secondary	7.01e-02	9.26e-07
Japanese Electric Grid	Xylene (mixed isomers)	model/secondary	6.95e-02	9.17e-07
LPG Production	Acetophenone	secondary	6.71e-02	8.85e-07
Natural Gas Prod.	Acrolein	secondary	4.89e-02	6.45e-07
Fuel Oil #6 Prod.	Toluene	secondary	4.18e-02	5.52e-07
Fuel Oil #6 Prod.	Carbon disulfide	secondary	3.91e-02	5.16e-07
Natural Gas Prod.	Benzyl chloride	secondary	3.88e-02	5.12e-07
Fuel Oil #4 Prod.	Benzene	secondary	3.74e-02	4.94e-07
LPG Production	Tetrachloroethylene	secondary	3.47e-02	4.59e-07
US electric grid	Chlorobenzene	model/secondary	3.16e-02	4.17e-07
Fuel Oil #6 Prod.	Methyl ethyl ketone	secondary	3.10e-02	4.10e-07
Fuel Oil #6 Prod.	Isophorone	secondary	2.85e-02	3.77e-07
US electric grid	Vinyl acetate	model/secondary	2.73e-02	3.60e-07
LPG Production	1,1,1-Trichloroethane	secondary	2.53e-02	3.34e-07
Japanese Electric Grid	Chlorobenzene	model/secondary	2.52e-02	3.33e-07
Japanese Electric Grid	o-xylene	model/secondary	2.48e-02	3.27e-07
LPG Production	Chlorine	secondary	2.32e-02	3.06e-07
Fuel Oil #6 Prod.	Phenol	secondary	2.22e-02	2.93e-07
Japanese Electric Grid	Vinyl acetate	model/secondary	2.17e-02	2.87e-07
Fuel Oil #6 Prod.	Styrene	secondary	1.99e-02	2.63e-07
Fuel Oil #2 Prod.	Toluene	secondary	1.89e-02	2.49e-07
US electric grid	Benzene	model/secondary	1.73e-02	2.28e-07
Fuel Oil #2 Prod.	Carbon disulfide	secondary	1.52e-02	2.01e-07
US electric grid	Acetophenone	model/secondary	1.43e-02	1.89e-07
Japanese Electric Grid	Benzene	model/secondary	1.41e-02	1.87e-07
Japanese Electric Grid	1,1,1-Trichloroethane	model/secondary	1.22e-02	1.61e-07
Fuel Oil #2 Prod.	Methyl ethyl ketone	secondary	1.21e-02	1.59e-07
Japanese Electric Grid	Acetophenone	model/secondary	1.15e-02	1.51e-07
Fuel Oil #2 Prod.	Isophorone	secondary	1.11e-02	1.46e-07
LPG Production	1,2-Dichloroethane	secondary	1.07e-02	1.42e-07
Fuel Oil #6 Prod.	Naphthalene	secondary	1.06e-02	1.40e-07
Fuel Oil #4 Prod.	Acrolein	secondary	9.43e-03	1.24e-07
Natural Gas Prod.	Naphthalene	secondary	8.83e-03	1.17e-07
Fuel Oil #2 Prod.	Phenol	secondary	8.63e-03	1.14e-07
Natural Gas Prod.	Carbon disulfide	secondary	8.40e-03	1.11e-07
Fuel Oil #6 Prod.	o-xylene	secondary	8.28e-03	1.09e-07
Fuel Oil #2 Prod.	Styrene	secondary	7.73e-03	1.02e-07

Process Group	Material	LCI Data Type	Odor (m3)	% of Total
Fuel Oil #4 Prod.	Benzyl chloride	secondary	7.48e-03	9.87e-08
US electric grid	Tetrachloroethylene	model/secondary	7.43e-03	9.81e-08
Fuel Oil #6 Prod.	Cumene	secondary	7.17e-03	9.46e-08
Natural Gas Prod.	Methyl ethyl ketone	secondary	6.67e-03	8.81e-08
Natural Gas Prod.	Isophorone	secondary	6.13e-03	8.10e-08
Japanese Electric Grid	Tetrachloroethylene	model/secondary	5.94e-03	7.83e-08
US electric grid	1,1,1-Trichloroethane	model/secondary	5.79e-03	7.64e-08
Fuel Oil #6 Prod.	Methyl methacrylate	secondary	5.41e-03	7.14e-08
Fuel Oil #2 Prod.	Naphthalene	secondary	5.24e-03	6.92e-08
Fuel Oil #6 Prod.	Methyl hydrazine	secondary	4.84e-03	6.39e-08
Natural Gas Prod.	Phenol	secondary	4.77e-03	6.30e-08
Natural Gas Prod.	Styrene	secondary	4.28e-03	5.65e-08
Fuel Oil #2 Prod.	o-xylene	secondary	3.99e-03	5.27e-08
Fuel Oil #6 Prod.	2-Chloroacetophenone	secondary	3.79e-03	5.00e-08
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	3.71e-03	4.90e-08
LPG Production	Dichloromethane	secondary	3.04e-03	4.01e-08
Fuel Oil #2 Prod.	Cumene	secondary	2.79e-03	3.68e-08
Fuel Oil #6 Prod.	1,4-Dichlorobenzene	secondary	2.36e-03	3.11e-08
US electric grid	1,2-Dichloroethane	model/secondary	2.30e-03	3.03e-08
Fuel Oil #2 Prod.	Methyl methacrylate	secondary	2.10e-03	2.78e-08
Natural Gas Prod.	Chlorine	secondary	2.06e-03	2.72e-08
Fuel Oil #4 Prod.	Toluene	secondary	1.89e-03	2.49e-08
Fuel Oil #2 Prod.	Methyl hydrazine	secondary	1.88e-03	2.48e-08
Japanese Electric Grid	1.2-Dichloroethane	model/secondary	1.83e-03	2.42e-08
Fuel Oil #4 Prod.	Carbon disulfide	secondary	1.62e-03	2.14e-08
Natural Gas Prod.	Cumene	secondary	1.54e-03	2.03e-08
Fuel Oil #2 Prod.	2-Chloroacetophenone	secondary	1.47e-03	1.94e-08
Fuel Oil #4 Prod.	Methyl ethyl ketone	secondary	1.29e-03	1.70e-08
Fuel Oil #2 Prod.	1.4-Dichlorobenzene	secondary	1.22e-03	1.62e-08
Fuel Oil #6 Prod.	Chlorobenzene	secondary	1.19e-03	1.57e-08
Fuel Oil #4 Prod.	Isophorone	secondary	1.18e-03	1.56e-08
US electric grid	o-xvlene	model/secondary	1.17e-03	1.55e-08
Natural Gas Prod.	Methyl methacrylate	secondary	1.16e-03	1.54e-08
Natural Gas Prod.	Methyl hydrazine	secondary	1.04e-03	1.37e-08
Fuel Oil #6 Prod.	Vinyl acetate	secondary	1.03e-03	1.36e-08
Fuel Oil #4 Prod.	Phenol	secondary	9.20e-04	1.21e-08
Fuel Oil #4 Prod	Styrene	secondary	8 25e-04	1.09e-08
Natural Gas Prod.	2-Chloroacetophenone	secondary	8.14e-04	1.07e-08
US electric orid	Dichloromethane	model/secondary	6 50e-04	8 58e-09
LPG Production	Chloroform	secondary	6 09e-04	8 04e-09
Fuel Oil #6 Prod	Acetophenone	secondary	5 41e-04	7 14e-09
Jananese Electric Grid	Dichloromethane	model/secondary	5 19e-04	6.85e-09
Fuel Oil #4 Prod	Nanhthalene	secondary	5.05e-04	6 67e-09
Fuel Oil #2 Prod	Chlorobenzene	secondary	4 63e-04	6 11e-09
Fuel Oil #2 Prod	Vinyl acetate	secondary	4 00e-04	5 28e-09
Fuel Oil #4 Prod	o-xylene	secondary	3 89e-04	5.13e-09
Fuel Oil #4 Prod	Cumene	secondary	2 97e-04	3.92e-09
Fuel Oil #6 Prod	Tetrachloroethylene	secondary	2.970-04	3.700-00
Natural Gas Prod	Chlorobenzene	secondary	2.000-04	3 380 00
Fuel Oil #4 Prod	Mathyl mathacrylate	secondary	2.500-04	2.060.00
Natural Gas Drod	Vinul agetete	secondary	2.240-04	2.900-09
Inatural Gas Prod.	v myt acetate	secondary	2.21e-04	2.92e-09

	SITA Results for the Resth	enes impacts Ca	itegory	
Process Group	Material	LCI Data Type	Odor (m3)	% of Total
Fuel Oil #2 Prod.	Acetophenone	secondary	2.10e-04	2.78e-09
Fuel Oil #6 Prod.	1,1,1-Trichloroethane	secondary	2.04e-04	2.70e-09
Fuel Oil #4 Prod.	Methyl hydrazine	secondary	2.01e-04	2.65e-09
Fuel Oil #4 Prod.	2-Chloroacetophenone	secondary	1.57e-04	2.07e-09
US electric grid	Chloroform	model/secondary	1.30e-04	1.72e-09
Fuel Oil #6 Prod.	Chlorine	secondary	1.29e-04	1.70e-09
Natural Gas Prod.	Acetophenone	secondary	1.16e-04	1.54e-09
Fuel Oil #4 Prod.	1,4-Dichlorobenzene	secondary	1.16e-04	1.53e-09
Fuel Oil #2 Prod.	Tetrachloroethylene	secondary	1.09e-04	1.44e-09
Japanese Electric Grid	Chloroform	model/secondary	1.04e-04	1.37e-09
Fuel Oil #6 Prod.	1,2-Dichloroethane	secondary	8.66e-05	1.14e-09
Fuel Oil #2 Prod.	1,1,1-Trichloroethane	secondary	7.94e-05	1.05e-09
Fuel Oil #2 Prod.	Chlorine	secondary	6.80e-05	8.97e-10
Natural Gas Prod.	Tetrachloroethylene	secondary	6.03e-05	7.96e-10
Fuel Oil #4 Prod.	Chlorobenzene	secondary	4.94e-05	6.52e-10
Natural Gas Prod.	1,1,1-Trichloroethane	secondary	4.39e-05	5.79e-10
Fuel Oil #4 Prod.	Vinvl acetate	secondary	4.26e-05	5.63e-10
Fuel Oil #2 Prod.	1.2-Dichloroethane	secondary	3.37e-05	4.44e-10
Fuel Oil #6 Prod	Dichloromethane	secondary	2 45e-05	3 24e-10
Fuel Oil #4 Prod	Acetophenone	secondary	2.13e 05	2.96e-10
Natural Gas Prod	1.2-Dichloroethane	secondary	1.86e-05	2.96e-10
Fuel Oil #4 Prod	Tetrachloroethylene	secondary	1.00e 05	1.53e-10
Fuel Oil #2 Prod	Dichloromethane	secondary	9.53e-06	1.35e 10
Fuel Oil #2 Prod	1 1 1-Trichloroethane	secondary	8.47e-06	1.20e 10
Fuel Oil #4 Prod	Chlorine	secondary	6.40e-06	8.44e-11
Natural Gas Prod	Dichloromethane	secondary	5 270 06	6.96e.11
Fuel Oil #6 Drod	Chloroform	secondary	5.27e-00	6.482.11
Fuel Oil #4 Prod	1.2 Disklangsthang	secondary	4.91e-00	4.740.11
Fuel Oil #2 Prod	Chloroform	secondary	1.010.06	4.746-11
Fuel OII #2 Flog.	Chloroform	secondary	1.916-00	2.32e-11
Natural Gas Prod.	Dishlawayathawa	secondary	1.000-00	1.39e-11
Fuel Oil #4 Prod.	Chlanafarm	secondary	2.0407	1.34e-11
Fuel Oli #4 Prod.	Chlorolorm	secondary	2.04e-07	2.09e-12
Total Manufacturing			7.41e+06	9.78e+01
Use, Maintenance and Repair Life-cycle St	age			
US electric grid	Acetaldehyde	model/secondary	1.90e+05	2.50e+00
US electric grid	Propionaldehyde	model/secondary	9.76e+03	1.29e-01
US electric grid	Acrolein	model/secondary	3.78e+02	4.99e-03
US electric grid	Benzyl chloride	model/secondary	3.00e+02	3.96e-03
US electric grid	Formaldehyde	model/secondary	1.73e+02	2.29e-03
US electric grid	Carbon disulfide	model/secondary	6.49e+01	8.57e-04
US electric grid	Methyl ethyl ketone	model/secondary	5.16e+01	6.81e-04
US electric grid	Isophorone	model/secondary	4.74e+01	6.26e-04
US electric grid	Toluene	model/secondary	4.25e+01	5.61e-04
US electric grid	Phenol	model/secondary	3.69e+01	4.87e-04
US electric grid	Stvrene	model/secondary	3.30e+01	4.36e-04
US electric grid	Cumene	model/secondary	1.19e+01	1.57e-04
US electric grid	Naphthalene	model/secondary	9.16e+00	1.21e-04
US electric grid	Methyl methacrylate	model/secondary	8.99e+00	1.19e-04
US electric grid	Methyl hydrazine	model/secondary	8.04e+00	1.06e-04
US electric grid	2-Chloroacetophenone	model/secondary	6.29e+0.0	8 31e-05
US electric grid	Xylene (mixed isomers)	model/secondary	5 45e+00	7.20e-05
	,	a a a a a a a a a a a a a a a a a	2.120.00	

Process Group	Material	LCI Data Type	Odor (m3)	% of Total
US electric grid	Chlorobenzene	model/secondary	1.98e+00	2.61e-05
US electric grid	Vinyl acetate	model/secondary	1.71e+00	2.25e-05
US electric grid	Benzene	model/secondary	1.08e+00	1.43e-05
US electric grid	Acetophenone	model/secondary	8.99e-01	1.19e-05
US electric grid	Tetrachloroethylene	model/secondary	4.66e-01	6.15e-06
US electric grid	1,1,1-Trichloroethane	model/secondary	3.63e-01	4.79e-06
US electric grid	1,2-Dichloroethane	model/secondary	1.44e-01	1.90e-06
US electric grid	o-xylene	model/secondary	7.33e-02	9.68e-07
US electric grid	Dichloromethane	model/secondary	4.07e-02	5.38e-07
US electric grid	Chloroform	model/secondary	8.16e-03	1.08e-07
Total Use, Maintenance and Repair			2.01e+05	2.65e+00
End-of-life Life-cycle Stage				
CRT landfilling	Hydrogen sulfide	primary	5.37e+02	7.09e-03
LPG Production	Hydrogen sulfide	secondary	6.17e+01	8.14e-04
US electric grid	Acetaldehyde	model/secondary	1 90e+01	2.51e-04
US electric grid	Propionaldehyde	model/secondary	9 76e-01	1 29e-05
LPG Production	Acetaldehyde	secondary	1 22e-01	1.61e-06
CRT landfilling	Toluene	nrimary	7.98e-02	1.05e-06
CRT landfilling	Xylene (mixed isomers)	primary	4.07e-02	5 37e-07
US electric grid	A crolein	model/secondary	3.78e-02	4 99e-07
US electric grid	Benzyl chloride	model/secondary	3.00e-02	3.96e-07
L PG Production	Formaldehyde	secondary	2.07e-02	2.73e-07
L PG Production	Ammonia	secondary	1.88e-02	2.75e-07
US electric grid	Formaldehyde	model/secondary	1.73e-02	2.19e-07
US electric grid	Carbon disulfide	model/secondary	6 50e 03	2.29C-07
L PG Production	Propionaldebyde	secondary	6 280 03	8.37c-08
CPT londfilling	Pangana	secondary	6,282-03	8.300-08
	Method etherl leaters	primary	5.1602	6.296-08
US electric grid	Isonhorene	model/secondary	3.100-03	6.260.08
US electric grid	Taluana	model/secondary	4.746-03	5.610.08
US electric grid	I oluelle	model/secondary	4.236-03	3.01e-08
US electric grid	Phenol Stamon a	model/secondary	3.096-03	4.8/e-08
	Styrene	model/secondary	3.316-03	4.366-08
US electric grid	Cumene	model/secondary	1.19e-03	1.57e-08
LPG Production	Benzene	secondary	1.15e-03	1.52e-08
US electric grid	Naphthalene	model/secondary	9.17e-04	1.21e-08
US electric grid	Methyl methacrylate	model/secondary	8.99e-04	1.19e-08
US electric grid	Methyl hydrazine	model/secondary	8.05e-04	1.06e-08
US electric grid	2-Chloroacetophenone	model/secondary	6.30e-04	8.31e-09
US electric grid	Xylene (mixed isomers)	model/secondary	5.45e-04	7.20e-09
CRT Incineration	Trichloroethylene	secondary	3.77e-04	4.98e-09
CRT landfilling	Trichloroethylene	primary	3.52e-04	4.64e-09
LPG Production	Acrolein	secondary	2.43e-04	3.21e-09
US electric grid	Chlorobenzene	model/secondary	1.98e-04	2.61e-09
LPG Production	Benzyl chloride	secondary	1.93e-04	2.55e-09
US electric grid	Vinyl acetate	model/secondary	1.71e-04	2.26e-09
CRT landfilling	Tetrachloroethylene	primary	1.65e-04	2.18e-09
US electric grid	Benzene	model/secondary	1.08e-04	1.43e-09
US electric grid	Acetophenone	model/secondary	8.99e-05	1.19e-09
CRT landfilling	1,2-Dichloroethane	primary	5.49e-05	7.25e-10
LPG Production	Toluene	secondary	5.36e-05	7.07e-10
Process Group	Material	LCI Data Type	Odor (m3)	% of Total
-------------------	-----------------------	-----------------	-----------	------------
US electric grid	Tetrachloroethylene	model/secondary	4.66e-05	6.15e-10
LPG Production	Carbon disulfide	secondary	4.18e-05	5.52e-10
US electric grid	1,1,1-Trichloroethane	model/secondary	3.63e-05	4.79e-10
LPG Production	Methyl ethyl ketone	secondary	3.32e-05	4.38e-10
LPG Production	Isophorone	secondary	3.05e-05	4.03e-10
LPG Production	Phenol	secondary	2.37e-05	3.13e-10
LPG Production	Styrene	secondary	2.13e-05	2.81e-10
LPG Production	Naphthalene	secondary	1.51e-05	1.99e-10
US electric grid	1,2-Dichloroethane	model/secondary	1.44e-05	1.90e-10
LPG Production	o-xylene	secondary	1.14e-05	1.51e-10
LPG Production	Cumene	secondary	7.67e-06	1.01e-10
US electric grid	o-xylene	model/secondary	7.34e-06	9.69e-11
LPG Production	Methyl methacrylate	secondary	5.79e-06	7.64e-11
LPG Production	Methyl hydrazine	secondary	5.18e-06	6.84e-11
US electric grid	Dichloromethane	model/secondary	4.08e-06	5.38e-11
LPG Production	2-Chloroacetophenone	secondary	4.05e-06	5.35e-11
LPG Production	1,4-Dichlorobenzene	secondary	3.56e-06	4.70e-11
CRT landfilling	Dichloromethane	primary	3.43e-06	4.52e-11
CRT landfilling	Chloroform	primary	2.11e-06	2.79e-11
CRT Incineration	Carbon tetrachloride	secondary	1.66e-06	2.20e-11
CRT landfilling	Carbon tetrachloride	primary	1.55e-06	2.05e-11
LPG Production	Chlorobenzene	secondary	1.27e-06	1.68e-11
LPG Production	Vinyl acetate	secondary	1.10e-06	1.45e-11
US electric grid	Chloroform	model/secondary	8.16e-07	1.08e-11
LPG Production	Acetophenone	secondary	5.79e-07	7.64e-12
LPG Production	Tetrachloroethylene	secondary	3.00e-07	3.96e-12
LPG Production	1,1,1-Trichloroethane	secondary	2.18e-07	2.88e-12
LPG Production	Chlorine	secondary	2.00e-07	2.64e-12
LPG Production	1,2-Dichloroethane	secondary	9.26e-08	1.22e-12
LPG Production	Dichloromethane	secondary	2.62e-08	3.46e-13
LPG Production	Chloroform	secondary	5.25e-09	6.94e-14
Natural Gas Prod.	Chloroform	secondary	-5.29e-07	-6.99e-12
Fuel Oil #4 Prod.	Chloroform	secondary	-2.19e-06	-2.89e-11
Natural Gas Prod.	Dichloromethane	secondary	-2.64e-06	-3.49e-11
Natural Gas Prod.	1,2-Dichloroethane	secondary	-9.33e-06	-1.23e-10
Fuel Oil #4 Prod.	Dichloromethane	secondary	-1.09e-05	-1.44e-10
Natural Gas Prod.	1,1,1-Trichloroethane	secondary	-2.20e-05	-2.91e-10
Natural Gas Prod.	Tetrachloroethylene	secondary	-3.02e-05	-3.99e-10
Fuel Oil #4 Prod.	1,2-Dichloroethane	secondary	-3.86e-05	-5.09e-10
Natural Gas Prod.	Acetophenone	secondary	-5.83e-05	-7.70e-10
Fuel Oil #4 Prod.	Chlorine	secondary	-6.88e-05	-9.08e-10
CRT Incineration	Chlorine	secondary	-8.44e-05	-1.11e-09
Fuel Oil #4 Prod.	1,1,1-Trichloroethane	secondary	-9.10e-05	-1.20e-09
Natural Gas Prod.	Vinyl acetate	secondary	-1.11e-04	-1.46e-09
Fuel Oil #4 Prod.	Tetrachloroethylene	secondary	-1.25e-04	-1.65e-09
Natural Gas Prod.	Chlorobenzene	secondary	-1.28e-04	-1.69e-09
CRT Incineration	Chloroform	secondary	-1.33e-04	-1.76e-09
Fuel Oil #4 Prod.	Acetophenone	secondary	-2.41e-04	-3.18e-09
Natural Gas Prod.	2-Chloroacetophenone	secondary	-4.08e-04	-5.39e-09
Fuel Oil #4 Prod.	Vinyl acetate	secondary	-4.58e-04	-6.05e-09
Natural Gas Prod.	Methyl hydrazine	secondary	-5.22e-04	-6.89e-09

Table M-35. CRT LCIA Results for the Aesthetics Impacts Category

Process Group	Material	LCI Data Type	Odor (m3)	% of Total
Fuel Oil #4 Prod.	Chlorobenzene	secondary	-5.31e-04	-7.00e-09
Natural Gas Prod.	Methyl methacrylate	secondary	-5.83e-04	-7.70e-09
CRT Incineration	Dichloromethane	secondary	-6.72e-04	-8.87e-09
Natural Gas Prod.	Cumene	secondary	-7.73e-04	-1.02e-08
Natural Gas Prod.	Chlorine	secondary	-1.03e-03	-1.37e-08
Fuel Oil #4 Prod.	1,4-Dichlorobenzene	secondary	-1.25e-03	-1.64e-08
Fuel Oil #4 Prod.	2-Chloroacetophenone	secondary	-1.69e-03	-2.23e-08
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	-1.86e-03	-2.46e-08
Natural Gas Prod.	Styrene	secondary	-2.14e-03	-2.83e-08
Fuel Oil #4 Prod.	Methyl hydrazine	secondary	-2.16e-03	-2.85e-08
CRT Incineration	1,2-Dichloroethane	secondary	-2.33e-03	-3.07e-08
Natural Gas Prod.	Phenol	secondary	-2.39e-03	-3.16e-08
Fuel Oil #4 Prod.	Methyl methacrylate	secondary	-2.41e-03	-3.18e-08
Natural Gas Prod.	Isophorone	secondary	-3.08e-03	-4.06e-08
Fuel Oil #4 Prod.	Cumene	secondary	-3.20e-03	-4.22e-08
Natural Gas Prod.	Methyl ethyl ketone	secondary	-3.35e-03	-4.42e-08
Fuel Oil #4 Prod.	o-xylene	secondary	-4.18e-03	-5.52e-08
Natural Gas Prod.	Carbon disulfide	secondary	-4.21e-03	-5.56e-08
Natural Gas Prod.	Naphthalene	secondary	-4.43e-03	-5.84e-08
Fuel Oil #4 Prod.	Naphthalene	secondary	-5.43e-03	-7.17e-08
CRT Incineration	1,1,1-Trichloroethane	secondary	-5.62e-03	-7.42e-08
CRT Incineration	Tetrachloroethylene	secondary	-7.54e-03	-9.96e-08
Fuel Oil #4 Prod.	Styrene	secondary	-8.87e-03	-1.17e-07
Fuel Oil #4 Prod.	Phenol	secondary	-9.89e-03	-1.31e-07
Fuel Oil #4 Prod.	Isophorone	secondary	-1.27e-02	-1.68e-07
Fuel Oil #4 Prod.	Methyl ethyl ketone	secondary	-1.38e-02	-1.83e-07
CRT Incineration	Acetophenone	secondary	-1.49e-02	-1.97e-07
Fuel Oil #4 Prod.	Carbon disulfide	secondary	-1.74e-02	-2.30e-07
Natural Gas Prod.	Benzyl chloride	secondary	-1.94e-02	-2.57e-07
Fuel Oil #4 Prod.	Toluene	secondary	-2.03e-02	-2.68e-07
Natural Gas Prod.	Acrolein	secondary	-2.45e-02	-3.24e-07
CRT Incineration	Vinyl acetate	secondary	-2.83e-02	-3.74e-07
CRT Incineration	Chlorobenzene	secondary	-3.28e-02	-4.33e-07
Natural Gas Prod.	Toluene	secondary	-4.33e-02	-5.71e-07
Natural Gas Prod.	o-xylene	secondary	-4.46e-02	-5.89e-07
CRT Incineration	Xylene (mixed isomers)	secondary	-5.26e-02	-6.95e-07
Fuel Oil #4 Prod.	Benzyl chloride	secondary	-8.04e-02	-1.06e-06
Fuel Oil #4 Prod.	Acrolein	secondary	-1.01e-01	-1.34e-06
CRT Incineration	Methyl hydrazine	secondary	-1.33e-01	-1.76e-06
CRT Incineration	Naphthalene	secondary	-1.38e-01	-1.82e-06
CRT Incineration	Methyl methacrylate	secondary	-1.49e-01	-1.97e-06
CRT Incineration	Cumene	secondary	-1.97e-01	-2.61e-06
Natural Gas Prod.	Formaldehyde	secondary	-2.48e-01	-3.28e-06
Fuel Oil #4 Prod.	Benzene	secondary	-4.02e-01	-5.31e-06
CRT Incineration	Benzene	secondary	-4.93e-01	-6.51e-06
CRT Incineration	Styrene	secondary	-5.48e-01	-7.23e-06
CRT Incineration	Phenol	secondary	-6.11e-01	-8.07e-06
Natural Gas Prod.	Propionaldehyde	secondary	-6.33e-01	-8.36e-06
CRT Incineration	Toluene	secondary	-6.67e-01	-8.81e-06
CRT Incineration	Isophorone	secondary	-7.86e-01	-1.04e-05
CRT Incineration	Methyl ethyl ketone	secondary	-8.55e-01	-1.13e-05

Table M-35. CRT LCIA Results for the Aesthetics Impacts Category

Process Group	Material	LCI Data Type	Odor (m3)	% of Total
CRT Incineration	Carbon disulfide	secondary	-1.08e+00	-1.42e-05
Fuel Oil #4 Prod.	Propionaldehyde	secondary	-2.62e+00	-3.46e-05
CRT Incineration	Benzyl chloride	secondary	-4.97e+00	-6.56e-05
Natural Gas Prod.	Benzene	secondary	-5.89e+00	-7.78e-05
CRT Incineration	Acrolein	secondary	-6.26e+00	-8.27e-05
Fuel Oil #4 Prod.	Ammonia	secondary	-6.41e+00	-8.47e-05
CRT Incineration	Formaldehyde	secondary	-7.98e+00	-1.05e-04
Fuel Oil #4 Prod.	Formaldehyde	secondary	-9.86e+00	-1.30e-04
Natural Gas Prod.	Acetaldehyde	secondary	-1.23e+01	-1.63e-04
CRT Incineration	Ammonia	secondary	-2.08e+01	-2.75e-04
Fuel Oil #4 Prod.	Acetaldehyde	secondary	-5.09e+01	-6.72e-04
Natural Gas Prod.	Hydrogen sulfide	secondary	-7.86e+01	-1.04e-03
Natural Gas Prod.	Ammonia	secondary	-7.87e+01	-1.04e-03
CRT Incineration	Propionaldehyde	secondary	-1.62e+02	-2.14e-03
CRT Incineration	Hydrogen sulfide	secondary	-1.42e+03	-1.88e-02
CRT Incineration	Acetaldehyde	secondary		-4.15e-02
Fuel Oil #4 Prod.	Hydrogen sulfide	secondary	-2.97e+04	-3.92e-01
Total End-of-life			-3.41e+04	-4.50e-01
Total All Life-cycle Stages			7.58e+06	1.00e+02

Table M-35. CRT LCIA Results for the Aesthetics Impacts Category

Process Group	Material	LCI Data Type	Odor (m3)	% of Total
Materials Processing Life-cycle Stage	e			
Steel Prod., cold-rolled, semi-finished	Hydrogen sulfide	secondary	1.55e+04	3.08e-01
Natural Gas Prod.	Ammonia	secondary	1.10e+04	2.17e-01
Natural Gas Prod.	Hydrogen sulfide	secondary	1.10e+04	2.17e-01
Polycarbonate Production	Ethanethiol	secondary	5.87e+03	1.16e-01
PMMA Sheet Prod.	Ethanethiol	secondary	4.37e+03	8.66e-02
PMMA Sheet Prod.	Hydrogen sulfide	secondary	3.58e+03	7.09e-02
Aluminum Prod.	Hydrogen sulfide	secondary	3.27e+03	6.48e-02
Polycarbonate Production	Hydrogen sulfide	secondary	2.40e+03	4.77e-02
Natural Gas Prod.	Acetaldehyde	secondary	1.72e+03	3.40e-02
Steel Prod., cold-rolled, semi-finished	Acetaldehyde	secondary	1.22e+03	2.41e-02
Natural Gas Prod.	Benzene	secondary	8.21e+02	1.63e-02
PET Resin Production	Hydrogen sulfide	secondary	2.12e+02	4.19e-03
Styrene-butadiene Copolymer Prod.	Hydrogen sulfide	secondary	2.11e+02	4.18e-03
Steel Prod., cold-rolled, semi-finished	Ammonia	secondary	2.03e+02	4.03e-03
Natural Gas Prod.	Propionaldehyde	secondary	8.82e+01	1.75e-03
Aluminum Prod.	Acetic acid	secondary	4.23e+01	8.38e-04
Natural Gas Prod.	Formaldehyde	secondary	3.46e+01	6.85e-04
Steel Prod., cold-rolled, semi-finished	Acetic acid	secondary	2.70e+01	5.35e-04
Steel Prod., cold-rolled, semi-finished	Phenol	secondary	2.36e+01	4.69e-04
Aluminum Prod.	Xylene (mixed isomers)	secondary	1.13e+01	2.24e-04
PMMA Sheet Prod.	Ammonia	secondary	8.07e+00	1.60e-04
Steel Prod., cold-rolled, semi-finished	Toluene	secondary	6.35e+00	1.26e-04
Natural Gas Prod.	o-xylene	secondary	6.21e+00	1.23e-04
Natural Gas Prod.	Toluene	secondary	6.03e+00	1.19e-04
Aluminum Prod.	Toluene	secondary	5.16e+00	1.02e-04
Steel Prod., cold-rolled, semi-finished	Xylene (mixed isomers)	secondary	4.40e+00	8.73e-05
Steel Prod., cold-rolled, semi-finished	Formaldehyde	secondary	3.63e+00	7.20e-05
Natural Gas Prod.	Acrolein	secondary	3.41e+00	6.77e-05
Aluminum Prod.	Ammonia	secondary	3.36e+00	6.66e-05
Natural Gas Prod.	Benzyl chloride	secondary	2.71e+00	5.37e-05
Polycarbonate Production	Chlorine	secondary	1.12e+00	2.23e-05
Steel Prod., cold-rolled, semi-finished	Ethanol	secondary	1.02e+00	2.02e-05
PMMA Sheet Prod.	Chlorine	secondary	8.36e-01	1.66e-05
Natural Gas Prod.	Naphthalene	secondary	6.17e-01	1.22e-05
Natural Gas Prod.	Carbon disulfide	secondary	5.87e-01	1.16e-05
Natural Gas Prod.	Methyl ethyl ketone	secondary	4.66e-01	9.24e-06
Natural Gas Prod.	Isophorone	secondary	4.28e-01	8.50e-06
PET Resin Production	Chlorine	secondary	3.95e-01	7.84e-06
Styrene-butadiene Copolymer Prod.	Chlorine	secondary	3.94e-01	7.81e-06
Natural Gas Prod.	Phenol	secondary	3.33e-01	6.61e-06
Natural Gas Prod.	Styrene	secondary	2.99e-01	5.92e-06
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	2.59e-01	5.14e-06
Steel Prod., cold-rolled, semi-finished	Methanol	secondary	2.02e-01	4.00e-06
Natural Gas Prod.	Chlorine	secondary	1.44e-01	2.86e-06
Steel Prod., cold-rolled, semi-finished	Benzene	secondary	1.27e-01	2.51e-06
Natural Gas Prod.	Cumene	secondary	1.08e-01	2.13e-06
PET Resin Production	Ammonia	secondary	9.10e-02	1.80e-06
Styrene-butadiene Copolymer Prod.	Ammonia	secondary	9.06e-02	1.80e-06
Natural Gas Prod.	Methyl methacrylate	secondary	8.13e-02	1.61e-06
Natural Gas Prod.	Methyl hydrazine	secondary	7.27e-02	1.44e-06

 Table M-36.
 LCD LCIA Results for the Aesthetics Impact Category

Natural Gas Prod. 2-Chloroacetophrone secondary 3.83-02 1.13-60 Steel Prod. ord-rolled, seni-finished Naphthalene secondary 3.65-02 7.34-60 Ahuminum Prod. Benzene secondary 1.75-02 4.35-67 Natural Gas Prod. Chlorobenzene secondary 1.75-02 3.56-07 Steel Prod. Acetophenone secondary 4.35-03 8.85-08 Natural Gas Prod. Terachoechythene secondary 4.30-03 6.65-03 Natural Gas Prod. 1.1.1-Frichloroethane secondary 3.00-03 6.65-03 Natural Gas Prod. 1.1.1-Frichloroethane secondary 3.00-03 6.65-04 Natural Gas Prod. 1.1.1-Frichloroethane secondary 3.00-03 6.65-04 Natural Gas Prod. Chloroforn secondary 3.00-03 6.65-06 Steel Prod. Chloroforn secondary 3.00-03 6.65-06 Steel Prod. Chloroforn secondary 3.00-03 6.65-06 Steel Prod. Chloroforn	Process Group	Material	LCI Data Type	Odor (m3)	% of Total
Aluminum Prod. Phenol secondary 3.82-02 7.88-002 Aluminum Prod. Benzene secondary 3.65-02 7.34-07 Aluminum Prod. Benzene secondary 1.70-02 3.54-07 Matural Gias Prod. Chlorobenzene secondary 8.13-03 1.61-07 Scale Prod., cold-rolled, semi-finished Acetrophenone secondary 4.53-03 8.85-03 Scale Prod., cold-rolled, semi-finished Acetrophenone secondary 3.00-03 2.58-08 Natural Gias Prod. 1.1.1-Trichlorochane secondary 3.10-03 2.58-08 Natural Gias Prod. 1.2.Dichlorochane secondary 3.10-04 6.69-08 Natural Gias Prod. 1.2.Dichlorochane secondary 3.10-04 6.16-00 Natural Gias Prod. Dichloromethane secondary 3.10-04 1.46-04 Scal Prod., cold-rolled, semi-finished Propional delyde secondary 3.18-04 1.42-00 Matural Gias Prod. Prod. Chlorofforn secondary 3.18-04 1.24-00	Natural Gas Prod.	2-Chloroacetophenone	secondary	5.69e-02	1.13e-06
Steel Prod., cold-rolled, semi-finished Naphthalene secondary 2.36-02 2.43-07 Natural Gas Prod. Chlorobenzene secondary 1.79-02 3.54-07 Natural Gas Prod. Vinyl acetate secondary 8.12-60 3.54-07 Steel Prod., cold-rolled, semi-finished Acetone secondary 4.21-60 8.55-03 8.99-08 Natural Gas Prod. Tarchalorenely Jene secondary 4.21-60 8.35-06 8.99-08 Natural Gas Prod. 1.1,1-Trichloroenshane secondary 1.30-64 6.16-09 Natural Gas Prod. Dichloromentane secondary 3.88-04 6.16-09 Natural Gas Prod. Dichloromentane secondary 3.88-04 6.16-09 Steel Prod., cold-rolled, semi-finished Chlorosofam secondary 3.78-04 6.16-09 Steel Prod., cold-rolled, semi-finished Chlorosofam secondary 3.78-04 7.78-04 Steel Prod., cold-rolled, semi-finished Chlorosofam secondary 3.78-04 7.42-07 Steel Prod., cold-rolled, semi-finished Chlorosofam	Aluminum Prod.	Phenol	secondary	3.82e-02	7.58e-07
Aluminum Prod. Benzene secondary 2.23e-02 4.43e-07 Natural Gas Prod. Chiovenezne secondary 1.54e-02 5.54e-07 Natural Gas Prod. Acetophenone secondary 4.55e-03 1.61e-07 Stell Prod., cold-rolled, semi-finished Acetophenone secondary 4.55e-03 6.53e-06 Natural Gas Prod. 1.1-Trichloroethane secondary 3.07e-03 6.58e-08 Natural Gas Prod. 1.2-Dickloroethane secondary 3.08e-04 7.00e-09 Stell Prod., cold-rolled, semi-finished Dichloromethane secondary 3.38e-04 7.00e-09 Stell Prod., cold-rolled, semi-finished Dichloromethane secondary 3.38e-05 7.00e-10 Stell Prod., cold-rolled, semi-finished Projoronldehyde secondary 3.38e-05 7.00e-10 Stell Prod., cold-rolled, semi-finished Projoronldehyde secondary 3.38e-06 8.58v+01 Drod Chioroform secondary 3.48e-06 8.58v+01 Drod Hydrogen sulfide secondary 3.48e-06 8.58v+01 </td <td>Steel Prod., cold-rolled, semi-finished</td> <td>Naphthalene</td> <td>secondary</td> <td>3.65e-02</td> <td>7.24e-07</td>	Steel Prod., cold-rolled, semi-finished	Naphthalene	secondary	3.65e-02	7.24e-07
Natural Gas Prod. Chlorobenzene secondary 1.34-02 2.34-07 Natural Gas Prod. Acetophenone secondary 8.13e-03 1.61e-07 Steel Prod. Tetchhorothylene secondary 4.21e-03 8.09e-08 Natural Gas Prod. Tetchhorothylene secondary 4.21e-03 8.09e-08 Natural Gas Prod. 1.1,1-Trichhorothyne secondary 3.07e-03 6.08e-06 Natural Gas Prod. 1.2,1-Trichhorothyne secondary 3.08e-04 7.30e-09 Steel Prod., cold-rollel, semi-finished Dichlorothane secondary 7.38e-05 7.70e-10 Steel Prod., cold-rolle, semi-finished Chlorotom secondary 3.78e-05 7.70e-10 Steel Prod., cold-rolle, semi-finished Chlorotom secondary 3.78e-05 7.70e-10 Steel Prod., cold-rolle, semi-finished Phopine primary 4.48e-06 8.08e+01 Manifer Mod., cold-rolle, semi-finished Phopine primary 4.38e-06 6.80e+00 Monifer Mod., cold-rolle, semi-finished Phopine primary 2.38e-01	Aluminum Prod.	Benzene	secondary	2.23e-02	4.43e-07
Natural Gas Prod. Vinyl acetanc secondary 8.13-6-07 Steel Prod., cold-rolled, semi-finished Acetone secondary 4.53-643 8.99-68 Natural Gas Prod. Ticrach/orectplene secondary 4.51-643 8.35-618 Natural Gas Prod. 1,1-Trich/orectmare secondary 3.07-64 6.08-608 Natural Gas Prod. 1,2-Dich/orocthane secondary 3.86-04 7.38-640 Steel Prod., cold-rolled, semi-finished Isopory/proglonate secondary 3.86-04 7.38-645 1.46-609 Steel Prod., cold-rolled, semi-finished Projonaldehyde secondary 3.38-64 1.46-609 Steel Prod., cold-rolled, semi-finished Projonaldehyde secondary 3.38-64 1.42-600 Steel Prod., cold-rolled, semi-finished Projonaldehyde secondary 3.38-648 1.42-640 Dial Materials Processing J.78-645 1.42-640 1.22-400 Manufacturing Life-cycle Stage Material primary 3.43-648 6.88-6400 Dial Materials Processing Alge-610 primary 3.43-6420	Natural Gas Prod.	Chlorobenzene	secondary	1.79e-02	3.54e-07
Natural Gas Prod. Actophenone secondary #.13-e03 1.61-e07 Stect Prod., cold-rolled, semi-finished Actone secondary 4.21e-03 8.35e-08 Natural Gas Prod. 1.2.1-Tirchikoroschnane secondary 1.30e-03 2.58e-08 Natural Gas Prod. 1.2.Dichikoroschnane secondary 3.68e-04 7.36e-05 Steel Prod., cold-rolled, semi-finished Sopropyl propionate secondary 3.78e-05 1.46e-09 Steel Prod., cold-rolled, semi-finished Chloroform secondary 3.78e-05 1.46e-09 Steel Prod., cold-rolled, semi-finished Chloroform secondary 3.78e-05 1.46e-09 Steel Prod., cold-rolled, semi-finished Phopinaldebyde secondary 3.78e-05 7.90e-10 Mattriats Processing	Natural Gas Prod.	Vinyl acetate	secondary	1.54e-02	3.06e-07
Steel Prod., cold-rolled, semi-finishedActonesecondary4.53-e038.99-e08Natural Gas Prod.1.1.1-Trichlorocthinesecondary3.07-e036.68e-68Natural Gas Prod.1.2.Dichlorocthinesecondary1.36e-042.58e-06Natural Gas Prod.Dichloromcthinesecondary3.16e-046.16e-09Natural Gas Prod.Chloroformsecondary3.16e-046.16e-09Natural Gas Prod.Chloroformsecondary3.98e-057.90e-10Steel Prod., cold-rolled, semi-finishedChloroformsecondary3.98e-057.90e-10Steel Prod., cold-rolled, semi-finishedPropionaldelydesecondary3.98e-057.90e-10Steel Prod., cold-rolled, semi-finishedPropionaldelydesecondary3.98e-057.90e-10Steel Prod., cold-rolled, semi-finishedPropionaldelydesecondary3.18e-148.18e-100Disteel Prod., cold-rolled, semi-finishedPropionaldelydesecondary4.18e-168.80e-10DirotirmoduleAmmoniaprimary6.24e-1041.24e-100Monitor/moduleAcetic acidprimary6.24e-1041.24e-100Monitor/moduleAcetic acidprimary6.24e-1041.24e-100Monitor/moduleAcetic acidprimary2.32e+144.42e-101Fuel Oil 14 Prod.Hydrogen sulfidesecondary4.15e+131.61e-01Fuel Oil 14 Prod.Hydrogen sulfidesecondary4.15e+135.00e-02Fuel Oil 14 Prod.Hydrogen sulfideseco	Natural Gas Prod.	Acetophenone	secondary	8.13e-03	1.61e-07
Natural Gas Prod. 1,1,1-Trichlorosethane secondary 3,07e-03 6,08e-08 Natural Gas Prod. 1,2-Dichlorosethane secondary 1,30e-03 2,58e-08 Natural Gas Prod. Dichloromethane secondary 3,68e-04 7,30e-03 Steel Prod., ocld-rolled, semi-finished Chloroform secondary 3,78e-05 1,46e-09 Steel Prod., ocld-rolled, semi-finished Chloroform secondary 3,78e-05 1,46e-09 Steel Prod., ocld-rolled, semi-finished Chloroform secondary 3,78e-05 1,22e+00 Mantarcturing Life-cycle Stage ////////////////////////////////////	Steel Prod., cold-rolled, semi-finished	Acetone	secondary	4.53e-03	8.99e-08
Natural Gas Prod. I,1-Trichthorcethane secondary 3.07e-03 6.08e-03 Natural Gas Prod. 1,2-Dichtorocthane secondary 1.30e-03 2.58e-08 Steel Prod., cold-rolled, semi-finished Isoprop) propionate secondary 7.38e-05 I.46e-09 Steel Prod., cold-rolled, semi-finished Chorino secondary 3.38e-05 7.98e-10 Steel Prod., cold-rolled, semi-finished Propionaldehyde secondary 3.38e-05 7.98e-10 Steel Prod., cold-rolled, semi-finished Propionaldehyde secondary 3.38e-05 7.99e-10 Steel Prod., cold-rolled, semi-finished Propionaldehyde secondary 3.48e-05 6.80e+100 Manufacturing Life-cycle Stage ////////////////////////////////////	Natural Gas Prod.	Tetrachloroethylene	secondary	4.21e-03	8.35e-08
Natural Gas Prod.1.30-6032.58-608Natural Gas Prod.Dichloromethanesecondary3.68-647.30-609Steel Prod., cold-rolled, semi-finishedIsopropylpropionatesecondary3.86-647.30-609Steel Prod., cold-rolled, semi-finishedChlorofornsecondary3.88-637.490-610Steel Prod., cold-rolled, semi-finishedPropionaldehydesecondary3.98-6457.90-610Steel Prod., cold-rolled, semi-finishedPropionaldehydesecondary3.98-6457.90-610Manufacturing Life-cycle Stagemonitor/moduleAmonita1.22-e4044.31-10Monitor/moduleManoniaprimary4.48-r066.88-r010Monitor/moduleAcetic acidprimary2.23-r044.42-c01Monitor/moduleAcetic acidprimary2.23-r044.42-c01Japanes Flextric GridAcetialdehydemodel/secondary8.13-r032.18-r02Fuel Oil # Prod.Hydrogen sulfidesecondary2.52-r035.00-c02Fuel Oil # Prod.Hydrogen sulfidesecondary3.67-r027.29-e03Statural Gas Prod.Hydrogen sulfidesecondary3.78-r037.29-e03 <td>Natural Gas Prod.</td> <td>1,1,1-Trichloroethane</td> <td>secondary</td> <td>3.07e-03</td> <td>6.08e-08</td>	Natural Gas Prod.	1,1,1-Trichloroethane	secondary	3.07e-03	6.08e-08
Natural Gas Prod.Dichloromethanesecondary3.68e-047.30e-09Steel Prod., cold-rolled, semi-finishedIsopropy/propionatesecondary3.10e-046.16e-09Steel Prod., cold-rolled, semi-finishedChloroformsecondary3.98e-057.90e-10Steel Prod., cold-rolled, semi-finishedPropionaldelydesecondary3.98e-057.90e-10Steel Prod., cold-rolled, semi-finishedPhosphineprimary4.18e+068.89e+01Dral Materials Processing	Natural Gas Prod.	1,2-Dichloroethane	secondary	1.30e-03	2.58e-08
Sted Prod., cold-rolled, semi-finishedIsopropylropionatesecondary3.10e-046.16e-09Natural Gas Prod., cold-rolled, semi-finishedChloroformsecondary3.28e-057.290e-10Sted Prod., cold-rolled, semi-finishedPropionaldehydesecondary2.17e-054.31e-10Total Materials Processingrole of the secondary3.48e-058.89e-01Manufacturing Life-cycle Stagemontor/moduleprimary4.48e-068.89e-01LPG ProductionHydrogen sulfidesecondary3.43e+056.80e+00Monitor/moduleAmmoniaprimary6.24e+041.24e+00Monitor/moduleAeetia acidprimary6.24e+041.24e+00Monitor/moduleAeetia acidprimary2.23e+044.42e-01Japanese Electric GridAeetia acidprimary2.23e+044.42e-01Leu Oil #A Prod.Hydrogen sulfidesecondary8.13e+031.61e-01Leu Oil #A Prod.Hydrogen sulfidesecondary1.10e+032.18e-02Leu Oil #A Prod.Hydrogen sulfidesecondary4.18e+028.30e-03Sted Fred, culturinAcetaldehydemodel/secondary4.18e+024.90e-03Natural Gas Prod.Hydrogen sulfidesecondary2.47e+024.90e-03Sted Fred, CulturinArmoniasecondary1.16e+022.90e-03Japanese Electric GridPropionaldehydemodel/secondary3.76e+022.90e-03Natural Gas Prod.Armoniasecondary1.56e+022.90e-03<	Natural Gas Prod.	Dichloromethane	secondary	3.68e-04	7.30e-09
Natural Gas Prod.Chloroformsecondary7.38e-057.90e-10Steel Prod., cold-rolled, semi-finishedPropionaldehydesecondary2.17e-054.31e-10Total Materials Processing6.16e+041.22e+00Monitor/modulePhosphineprimary4.48e+068.89e+01LPG ProductionHydrogen sulfidesecondary3.43e+056.80e+00Monitor/moduleAcetia acidprimary6.24e+041.24e+00Monitor/moduleAcetia acidprimary6.24e+044.42e+00Monitor/moduleAcetia acidprimary2.23e+044.42e-01Japanese Electric GridAcetialdehydemodel/secondary4.36e+031.61e-01Fuel Oil 6 Prod.Hydrogen sulfidesecondary2.52e+035.00e-02Fuel Oil 6 Prod.Hydrogen sulfidesecondary2.52e+035.00e-02Fuel Oil 6 Prod.Hydrogen sulfidesecondary3.67e+027.29e-03Japanese Electric GridAcetaldehydemodel/secondary3.67e+027.29e-03Sultarul Gas Prod.Acetaldehydemodel/secondary3.67e+027.29e-03Japanese Electric GridAcetaldehydesecondary2.47e+024.91e-03Japanese Electric GridAcetaldehydesecondary2.47e+024.91e-03Japanese Electric GridAcetaldehydesecondary3.67e+027.29e-03Japanese Electric GridArmoniasecondary3.67e+022.28e-03LPG ProductionFormaldehydesecondary3.	Steel Prod., cold-rolled, semi-finished	Isopropylpropionate	secondary	3.10e-04	6.16e-09
Steel Prod., cold-rolled, semi-finishedChlorinesecondary3.98e-057.90e-10Steel Prod., cold-rolled, semi-finishedPropionaldehydesecondary2.17e-054.31e-10 Manufacturing Life-cycle Stage 1.22e+00Monitor/modulePhosphineprimary4.48e+066.80e+00Monitor/moduleAmmoniaprimary2.34e+056.80e+00Monitor/moduleAmmoniaprimary2.32e+044.42e+01Japanese Electric GridAcetia caidprimary2.32e+044.42e+01Japanese Electric GridAcetaldehydemodel/secondary8.13e+031.61e-01Fuel Oil #A Prod.Hydrogen sulfidesecondary4.26e+038.44e+02LFO ProductionAcetaldehydesecondary2.52e+035.00e+02Fuel Oil #2 Prod.Hydrogen sulfidesecondary6.76e+027.29e-03Japanese Electric GridPropionaldehydemodel/secondary4.18e+028.30e-03US electric gridAcetaldehydemodel/secondary4.18e+024.91e-03Japanese Electric GridPropionaldehydesecondary2.47e+024.90e-03LFO ProductionArmoniasecondary2.47e+024.90e-03LFO ProductionFormaldehydesecondary1.56e+022.28e-03LFO ProductionFormaldehydemodel/secondary1.85e+011.80e-03Parter gridAcetaldehydemodel/secondary1.56e+022.88e-03Japanese Electric GridPromaldehyde	Natural Gas Prod.	Chloroform	secondary	7.38e-05	1.46e-09
Steel Prod., cold-rolled, semi-finished Propionaldehyde secondary 2.17c-05 4.31c-10 Total Materials Processing	Steel Prod., cold-rolled, semi-finished	Chlorine	secondary	3.98e-05	7.90e-10
Total Materials Processing6.16+041.22e+00Manufacturing Life-cycle StageMonitor/modulePhosphineprimary4.48e+068.89e+01LPG ProductionHydrogen sulfidesecondary3.43e+056.80e+00Monitor/moduleAmonoiaprimary6.24e+041.24e+00Monitor/moduleAcetic acidprimary2.23e+044.42e+03Japanese Electric GridAcetic acidprimary2.12e+038.14e+02Fuel Oil #4 Prod.Hydrogen sulfidesecondary2.52e+035.00e+02Leu Oil #2 Prod.Hydrogen sulfidesecondary1.10e+012.18e+02LPG ProductionAcetaldehydemodel/secondary1.10e+032.18e+02LPG ProductionAcetaldehydemodel/secondary4.76e+027.29e+03Natural Gas Prod.Ammoniasecondary2.47e+024.90e+03Natural Gas Prod.Hydrogen sulfidesecondary2.47e+024.90e+03LPG ProductionFormaldehydesecondary1.15e+022.28e+03Natural Gas Prod.Ammoniasecondary1.05e+022.08e+03LPG ProductionAmmoniasecondary1.05e+022.08e+03LPG ProductionAmmoniasecondary1.05e+022.08e+03LPG ProductionAmmoniasecondary1.05e+022.08e+03LPG ProductionAmmoniasecondary1.05e+022.08e+03LPG ProductionAmmoniasecondary1.05e+022.08e+03LPG Production <td>Steel Prod., cold-rolled, semi-finished</td> <td>Propionaldehyde</td> <td>secondary</td> <td>2.17e-05</td> <td>4.31e-10</td>	Steel Prod., cold-rolled, semi-finished	Propionaldehyde	secondary	2.17e-05	4.31e-10
Manufacturing Life-cycle Stage Monitor/module Phosphine primary 4.48e+06 8.89e+01 LPG Production Hydrogen sulfide secondary 3.43e+05 6.80e+00 Monitor/module Ammonia primary 6.24e+04 1.24e+00 Monitor/module Acetic acid primary 6.23e+04 4.42e-01 Japaness Electric Grid Acetaldehyde model/secondary 8.13e+03 1.61e-01 Juel Oil # Prod. Hydrogen sulfide secondary 2.52e+03 5.00e-02 Fuel Oil #0 Frod. Hydrogen sulfide secondary 1.10e+03 2.18e-02 Japaness Electric Grid Propionaldehyde model/secondary 4.18e+02 8.30e-03 Japaness Electric Grid Propionaldehyde model/secondary 4.18e+02 8.30e-03 Matural Gas Prod. Ammonia secondary 2.47e+02 4.90e-03 LPG Production Formaldehyde secondary 2.47e+02 4.90e-03 LPG Production Formaldehyde secondary 1.15e+02 2.28e-03	Total Materials Processing			6.16e+04	1.22e+00
MainterformedulePhosphineprimary4.48e+068.89e+01LPG ProductionHydrogen sulfidesecondary3.43e+056.80e+00Monitor/moduleArmoniaprimary6.24e+041.24e+00Monitor/moduleAcetia acidprimary2.23e+044.42e-01Japanese Electric GridAcetia acidprimary2.23e+044.42e-01Japanese Electric GridAcetaldehydemodel/secondary8.13e+031.61e-01Fuel Oil # Prod.Hydrogen sulfidesecondary2.52e+035.00e-02Fuel Oil # Prod.Hydrogen sulfidesecondary1.10e+032.18e-02LPG ProductionAcetaldehydesecondary6.79e+021.35e-02Japanese Electric GridPropionaldehydemodel/secondary3.67e+027.29e-03Matural Gas Prod.Ammoniasecondary2.47e+024.90e-03LPG ProductionFormaldehydesecondary1.15e+022.28e-03Natural Gas Prod.Hydrogen sulfidesecondary1.15e+022.08e-03LPG ProductionFormaldehydemodel/secondary7.02e+011.39e-03LPG ProductionArmoniasecondary7.02e+011.39e-03LPG ProductionArmoniasecondary1.05e+022.08e-03Panel componentsTolueneprimary9.09e+1011.80e-03Japanese Electric GridFormaldehydemodel/secondary3.78e+017.88e+04Japanese Electric GridFormaldehydesecondary3.78e+013.7	Manufacturing Life-cycle Stage				
Monton/InductionHydrogen sulfideprimary1.182+036.608-00Monitor/moduleAcetic acidprimary6.24e+041.24e+00Monitor/moduleAcetic acidprimary2.23+044.42e-01Japanese Electric GiridAcetaldehydemodel/secondary8.13e+031.61e-01Fuel Oil #4 Prod.Hydrogen sulfidesecondary2.52e+035.00e-02Fuel Oil #2 Prod.Hydrogen sulfidesecondary2.52e+035.00e-02Fuel Oil #2 Prod.Hydrogen sulfidesecondary6.79e+021.35e-02Japanese Electric GiridPropionaldehydemodel/secondary4.18e+028.30e-03US electric giridAcetaldehydemodel/secondary3.67e+027.29e-03Natural Gas Prod.Hydrogen sulfidesecondary2.47e+024.91e-03Natural Gas Prod.Hydrogen sulfidesecondary2.47e+024.91e-03LPG ProductionFormaldehydesecondary2.47e+022.80e-03LPG ProductionFormaldehydesecondary1.05e+022.80e-03LPG ProductionFormaldehydesecondary1.05e+022.80e-03LPG ProductionFormaldehydemodel/secondary7.24e+011.48e-03Japanese Electric GridFormaldehydemodel/secondary3.50e+016.94e-04LPG ProductionPropionaldehydemodel/secondary3.50e+016.94e-04Japanese Electric GridFormaldehydemodel/secondary3.50e+016.94e-04LPG Production	Monitor/module	Phosphine	primary	4 48e+06	8 89e+01
In StructurianTryongen annocDecodeMonitor/moduleArmoniaprimary5.24e+041.24e+00Japanese Electric GridAcetic acidprimary2.23e+044.42e-01Japanese Electric GridAcetaldehydemodel/secondary8.13e+031.61e-01Fuel Oil #0 Prod.Hydrogen sulfidesecondary2.52e+035.00e-02Fuel Oil #0 Prod.Hydrogen sulfidesecondary2.52e+035.00e-02Fuel Oil #0 Prod.Hydrogen sulfidesecondary6.79e+021.35e-02Japanese Electric GridPropionaldehydemodel/secondary4.18e+028.30e-03LPG ProductionAcetaldehydemodel/secondary3.67e+027.29e-03Natural Gas Prod.Hydrogen sulfidesecondary2.47e+024.91e-03LPG ProductionFormaldehydesecondary1.15e+022.28e-03LPG ProductionFormaldehydesecondary1.15e+022.28e-03LPG ProductionFormaldehydesecondary1.15e+022.28e-03LPG ProductionArmoniasecondary1.05e+022.08e-03LPG ProductionArmoniasecondary1.05e+022.08e-03LPG ProductionArmoniasecondary7.24e+011.44e-03Japanese Electric GridFormaldehydemodel/secondary3.87e+017.68e-04WB Mfg.Formaldehydemodel/secondary3.50e+016.93e-04US electric GridAcetaldehydesecondary3.50e+013.21e-04Japanes	LPG Production	Hydrogen sulfide	secondary	3.43e+05	6.80e+00
NontrovinduiteInitiationprimary0.2 et of1.740 toJapanese Electric GridAcetia acidprimary2.2 ster044.42e-01Fuel Oil # Prod.Hydrogen sulfidesecondary4.26e+038.44e-02Fuel Oil # Prod.Hydrogen sulfidesecondary2.52e+032.18e-02Fuel Oil # Prod.Hydrogen sulfidesecondary1.10e+032.18e-02LPG ProductionAcetaldehydesecondary6.79e+021.35e-02Japanese Electric GridPropionaldehydemodel/secondary4.18e+028.30e-03Natural Gas Prod.Acetaldehydemodel/secondary2.47e+024.91e-03Natural Gas Prod.Hydrogen sulfidesecondary2.47e+024.91e-03Natural Gas Prod.Hydrogen sulfidesecondary2.47e+024.90e-03LPG ProductionFormaldehydesecondary1.15e+022.28e-03LPG ProductionArmoniasecondary1.15e+022.28e-03Panel componentsTolueneprimary9.09e+011.80e-03BacklightEthanolprimary7.24e+011.44e-03Japanese Electric GridFormaldehydemodel/secondary3.57e+01Vas electric GridPropionaldehydesecondary3.49e+016.93e-04US electric GridAcetaldehydemodel/secondary3.49e+016.93e-04Vus electric GridPropionaldehydesecondary3.49e+016.93e-04Vus electric GridAcetaldehydesecondary3.49e+013	Monitor/module	Ammonia	primary	6 24e+04	1.24e+00
Japanese Electric GridActeatalehydemodel/secondary8.15e-011.16-01Fuel Oil #4 Prod.Hydrogen sulfidesecondary4.26e+038.44e-02Fuel Oil #2 Prod.Hydrogen sulfidesecondary2.52e+035.00e-02Fuel Oil #2 Prod.Hydrogen sulfidesecondary1.10e+032.18e-02LPG ProductionAcetaldehydesecondary6.79e+021.35e-02Japanese Electric GridPropionaldehydemodel/secondary3.67e+027.29e-03Natural Gas Prod.Ammoniasecondary2.47e+024.91e-03LPG ProductionFormaldehydesecondary2.47e+024.90e-03LPG ProductionFormaldehydesecondary1.05e+022.28e-03Parel ComponentsFolueneprimary9.09e+011.80e-03LPG ProductionAmmoniasecondary1.05e+022.28e-03Panel componentsTolueneprimary9.09e+011.80e-03Japanese Electric GridFormaldehydemodel/secondary3.50e+01Japanese Electric GridFormaldehydemodel/secondary3.50e+01Natural Gas Prod.Acetaldehydesecondary3.50e+016.93e-04US electric gridPropionaldehydemodel/secondary3.50e+016.93e-04US electric GridPropionaldehydesecondary3.50e+016.93e-04US electric GridPropionaldehydesecondary3.50e+013.57e-04Japanese Electric GridPropionaldehydesecondary3.69e+01<	Monitor/module	A cetic acid	primary	2 23e+04	4 42e-01
Apparese Electric GridTechnic Hydrogen sulfidesecondary6.75e-038.44e-02Fuel Oil #0 Prod.Hydrogen sulfidesecondary2.52e+035.00e-02Fuel Oil #2 Prod.Hydrogen sulfidesecondary1.10e+032.18e-02LPG ProductionAcetaldehydesecondary6.79e+021.35e-02Japanese Electric GridPropionaldehydemodel/secondary3.67e+027.29e-03Natural Gas Prod.Ammoniasecondary2.47e+024.91e-03Natural Gas Prod.Hydrogen sulfidesecondary2.47e+024.90e-03LPG ProductionFormaldehydesecondary1.15e+022.28e-03LPG ProductionFormaldehydesecondary1.05e+022.08e-03LPG ProductionFormaldehydemodel/secondary7.24e+011.44e-03Japanese Electric GridFormaldehydemodel/secondary7.02e+011.39e-03Japanese Electric GridFormaldehydesecondary3.50e-016.94e-04LPG ProductionPromaldehydesecondary3.50e-016.94e-04Japanese Electric GridFormaldehydemodel/secondary3.50e-016.94e-04LPG ProductionPropionaldehydesecondary3.50e-016.94e-04LPG ProductionPropionaldehydesecondary3.50e-016.94e-04LPG ProductionPropionaldehydesecondary3.50e-016.94e-04LPG ProductionPropionaldehydesecondary3.50e-016.94e-04LPG ProductionBenzene	Japanese Electric Grid	Acetaldehyde	model/secondary	8 13e+03	1.61e-01
Tote of it / TrotaTry bright and brief of an it / trotaTrotaTrotaFuel Oil #P Prod.Hydrogen sulfidesecondary2.52e+035.50e-02LPG ProductionAcetaldehydesecondary6.79e+021.35e-02Japanese Electric GridPropionaldehydemodel/secondary4.18e+028.30e-03Natural Gas Prod.Acetaldehydemodel/secondary3.67e+027.29e-03Natural Gas Prod.Hydrogen sulfidesecondary2.47e+024.91e-03LPG ProductionFormaldehydesecondary1.15e+022.28e-03LPG ProductionFormaldehydesecondary1.05e+022.28e-03LPG ProductionAmmoniasecondary1.05e+022.08e-03BacklightTolueneprimary9.09e+011.80e-03Japanese Electric GridFormaldehydemodel/secondary7.02e+011.44e-03Japanese Tolueneprimary7.02e+011.39e-03Natural Gas Prod.Acetaldehydemodel/secondary3.67e+04LPG ProductionPropionaldehydemodel/secondary3.67e+04LPG ProductionPropionaldehydemodel/secondary3.29e+01Natural Gas Prod.Benzenesecondary1.82e+013.27e-04Japanese Electric GridPropionaldehydemodel/secondary1.82e+013.27e-04Natural Gas Prod.Benzenesecondary1.82e+013.27e-04Japanese Electric GridAcetaldehydesecondary1.82e+013.27e-04Natural Gas P	Fuel Oil #4 Prod	Hydrogen sulfide	secondary	4 26e+03	8 44e-02
Tote of the Tool.Try begin statuteSecondary1.10e+032.12e+03LPG ProductionAcetaldehydesecondary6.79e+021.35e-02Japanese Electric GridPropionaldehydemodel/secondary4.18e+028.30e-03US electric gridAcetaldehydemodel/secondary3.67e+027.29e-03Natural Gas Prod.Ammoniasecondary2.47e+024.91e-03LPG ProductionFormaldehydesecondary2.47e+024.90e-03LPG ProductionFormaldehydesecondary1.15e+022.28e-03LPG ProductionAmmoniasecondary1.05e+022.08e-03Panel componentsTolueneprimary9.09e+011.80e-03BacklightEthanolprimary7.24e+011.44e-03Japanese Electric GridFormaldehydemodel/secondary7.02e+01Natural Gas Prod.Acetaldehydesecondary3.87e+017.68e-04WB Mfg.Formaldehydemodel/secondary3.50e+016.94e-04US electric GridPropionaldehydemodel/secondary3.87e+013.75e-04Japanese Electric GridAcroleinmodel/secondary1.88e+013.75e-04Japanese Electric GridAcroleinmodel/secondary1.82e+013.75e-04Japanese Electric GridAcroleinmodel/secondary1.82e+013.75e-04Japanese Electric GridAcroleinmodel/secondary1.82e+013.75e-04Japanese Electric GridAcroleinmodel/secondary1.78e+04<	Fuel Oil #6 Prod	Hydrogen sulfide	secondary	2 52e+03	5.00e-02
Les of MartiniaDiscussionDiscussionDiscussionLPG ProductionAcetaldehydesecondary4.18e+028.30e-03US electric gridAcetaldehydemodel/secondary3.67e+027.29e-03Natural Gas Prod.Ammoniasecondary2.47e+024.91e-03Natural Gas Prod.Hydrogen sulfidesecondary2.47e+022.98e-03LPG ProductionFormaldehydesecondary1.15e+022.28e-03LPG ProductionFormaldehydesecondary1.05e+022.08e-03LPG ProductionAmmoniasecondary1.05e+022.08e-03Panel componentsTolueneprimary9.09e+011.80e-03BacklightEthanolprimary7.02e+011.39e-03Japanese Electric GridFormaldehydemodel/secondary7.02e+011.39e-03Natural Gas Prod.Acetaldehydesecondary3.50e+016.94e-04LPG ProductionPropionaldehydemodel/secondary3.50e+016.93e-04US electric gridPropionaldehydemodel/secondary1.89e+013.57e-04Natural Gas Prod.Benzenesecondary1.82e+013.67e-04Japanese Electric GridAcroleinmodel/secondary1.82e+013.57e-04Japanese Electric GridAcroleinmodel/secondary1.52e+013.52e-04Japanese Electric GridAcroleinmodel/secondary1.28e+012.55e-04Japanese Electric GridAcetaldehydesecondary3.89e+001.59e-04 <td>Fuel Oil #2 Prod</td> <td>Hydrogen sulfide</td> <td>secondary</td> <td>1 10e+03</td> <td>2 18e-02</td>	Fuel Oil #2 Prod	Hydrogen sulfide	secondary	1 10e+03	2 18e-02
Arbeiteric GridPropionaldehydemodel/secondary4.18e+028.30e-03US electric gridAcetaldehydemodel/secondary3.67e+027.29e-03Natural Gas Prod.Hydrogen sulfidesecondary2.47e+024.91e-03LPG ProductionFormaldehydesecondary2.47e+024.90e-03LPG ProductionFormaldehydesecondary1.15e+022.28e-03Panel componentsTolueneprimary9.09e+011.80e-03BacklightEthanolprimary7.24e+011.44e-03Japanese Electric GridFormaldehydesecondary7.02e+011.39e-03Natural Gas Prod.Acetaldehydesecondary3.87e+017.68e-04PWB Mfg.Formaldehydesecondary3.50e+016.94e-04US electric gridPropionaldehydemodel/secondary3.49e+016.93e-04US electric gridPropionaldehydemodel/secondary1.5e+013.75e-04Natural Gas Prod.Benzenesecondary1.88e+013.75e-04Vatural Gas Prod.Benzenesecondary1.82e+013.21e-04Japanese Electric GridAcroleinmodel/secondary1.82e+013.21e-04Japanese Electric GridBenzenesecondary1.52e+013.21e-04Japanese Electric GridAcetaldehydesecondary1.52e+013.21e-04Japanese Electric GridBenzenesecondary3.89e+007.71e-05Japanese Electric GridNethylelesecondary3.89e+007.	LPG Production	Acetaldehyde	secondary	6 79e+02	1 35e-02
Departer drifterAccelaldehydemodel/secondary3.67e+027.29e-03Natural Gas Prod.Ammoniasecondary2.47e+024.91e-03Natural Gas Prod.Hydrogen sulfidesecondary2.47e+024.90e-03LPG ProductionFormaldehydesecondary1.15e+022.28e-03LPG ProductionAmmoniasecondary1.05e+022.08e-03Panel componentsTolueneprimary9.09e+011.80e-03BacklightEthanolprimary7.24e+011.44e-03Japanese Electric GridFormaldehydemodel/secondary7.02e+011.39e-03Natural Gas Prod.Accetaldehydesecondary3.87e+017.68e-04PWB Mfg.Formaldehydemodel/secondary3.50e+016.94e-04LPG ProductionPropionaldehydesecondary1.89e+013.75e-04VB electric gridPropionaldehydemodel/secondary1.89e+013.75e-04Japanese Electric GridBenzenesecondary1.82e+013.75e-04Japanese Electric GridBenzenesecondary1.82e+013.75e-04Japanese Electric GridBenzyl chloridemodel/secondary1.28e+013.75e-04Japanese Electric GridBenzyl chloridemodel/secondary1.28e+013.21e-04Japanese Electric GridBenzyl chloridemodel/secondary1.28e+013.21e-04Japanese Electric GridBenzyl chloridesecondary3.39e+001.59e-04Fuel Oil #A Prod.Acetaldehydes	Japanese Electric Grid	Propionaldehyde	model/secondary	4 18e+02	8 30e-03
Construct gridInteractivityInteractivityInteractivityNatural Gas Prod.Hydrogen sulfidesecondary2.47e+024.90e-03LPG ProductionFormaldehydesecondary1.15e+022.28e-03LPG ProductionAmmoniasecondary1.05e+022.08e-03Panel componentsTolueneprimary9.09e+011.80e-03BacklightEthanolprimary7.24e+011.44e-03Japanese Electric GridFormaldehydemodel/secondary7.02e+011.39e-03Natural Gas Prod.Acetaldehydesecondary3.87e+017.68e-04PWB Mfg.Formaldehydemodel/secondary3.50e+016.94e-04LPG ProductionPropionaldehydesecondary1.89e+013.75e-04Natural Gas Prod.Benzenesecondary1.88e+013.75e-04Japanese Electric GridBenzenesecondary1.28e+013.67e-04Japanese Electric GridBenzenesecondary1.28e+013.21e-04Japanese Electric GridBenzenesecondary1.28e+013.67e-04Japanese Electric GridBenzyl chloridemodel/secondary1.28e+013.21e-04Japanese Electric GridBenzyl chloridemodel/secondary1.28e+013.50e+001.59e-04LPG ProductionBenzyl chloridemodel/secondary8.03e+001.59e-041.59e-041.49e-041.59e-041.45e-041.59e-041.45e-041.59e-041.45e-041.59e-041.59e-041.45e-041.	US electric grid	Acetaldehyde	model/secondary	3 67e+02	7 29e-03
Natural Gas Prod.Hydrogen sulfidesecondary2.47e+024.90e-03LPG ProductionFormaldehydesecondary1.15e+022.28e-03LPG ProductionAmmoniasecondary1.05e+022.08e-03Panel componentsTolueneprimary9.09e+011.80e-03BacklightEthanolprimary7.24e+011.44e-03Japanese Electric GridFormaldehydemodel/secondary7.02e+011.39e-03Natural Gas Prod.Acetaldehydesecondary3.87e+017.68e-04PWB Mfg.Formaldehydemodel/secondary3.50e+016.94e-04LPG ProductionPropionaldehydesecondary3.49e+016.93e-04US electric gridPropionaldehydemodel/secondary1.85e+1013.67e-04Japanese Electric GridAcroleinmodel/secondary1.85e+1013.67e-04Japanese Electric GridAcroleinmodel/secondary1.85e+1013.67e-04Japanese Electric GridBenzenesecondary1.28e+012.55e-04Japanese Electric GridBenzenesecondary1.28e+012.55e-04Japanese Electric GridToluenemodel/secondary3.89e+001.59e-04Fuel Oil #4 Prod.Acetaldehydesecondary3.89e+001.27e-04Fuel Oil #6 Prod.Acetaldehydesecondary3.89e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.78e+005.52e-05Japanese Electric GridNaphthalenemodel/secondary<	Natural Gas Prod	Ammonia	secondary	2.47e+02	4 91e-03
LPG ProductionFormaldehydesecondary1.15e+022.28e-03LPG ProductionAmmoniasecondary1.05e+022.08e-03Panel componentsTolueneprimary9.09e+011.80e-03BacklightEthanolprimary7.24e+011.44e-03Japanese Electric GridFormaldehydemodel/secondary7.02e+011.39e-03Natural Gas Prod.Acetaldehydesecondary3.87e+017.68e-04PWB Mfg.Formaldehydemodel/secondary3.50e+016.94e-04LPG ProductionPropionaldehydesecondary3.49e+016.94e-04US electric gridPropionaldehydesecondary1.85e+013.75e-04Natural Gas Prod.Benzenesecondary1.85e+013.27e-04Japanese Electric GridAcroleinmodel/secondary1.62e+013.21e-04Japanese Electric GridBenzenesecondary1.28e+013.21e-04Japanese Electric GridBenzyl chloridemodel/secondary1.52e+013.21e-04Japanese Electric GridToluenemodel/secondary1.52e+013.21e-04Japanese Electric GridAcetaldehydesecondary3.89e+001.72e-04Fuel Oil #4 Prod.Acetaldehydesecondary3.89e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.89e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.78e+005.52e-05Japanese Electric GridNaphthalenemodel/second	Natural Gas Prod.	Hydrogen sulfide	secondary	2.47e+02	4.90e-03
LPG ProductionAmmoniasecondary1.05e+022.08e-03Panel componentsTolueneprimary9.09e+011.80e-03BacklightEthanolprimary7.24e+011.44e-03Japanese Electric GridFormaldehydemodel/secondary7.02e+011.39e-03Natural Gas Prod.Acetaldehydesecondary3.87e+017.68e-04PWB Mfg.Formaldehydemodel/secondary3.50e+016.94e-04LPG ProductionPropionaldehydesecondary3.50e+016.94e-04US electric gridPropionaldehydemodel/secondary3.49e+016.93e-04US electric gridPropionaldehydemodel/secondary1.85e+013.75e-04Natural Gas Prod.Benzenesecondary1.85e+013.67e-04Japanese Electric GridAcroleinmodel/secondary1.62e+013.21e-04Japanese Electric GridBenzyl chloridemodel/secondary1.28e+012.55e-04Japanese Electric GridToluenemodel/secondary7.30e+001.45e-04LPG ProductionBenzenesecondary3.39e+001.27e-04Fuel Oil #4 Prod.Acetaldehydesecondary3.89e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+007.30e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+007.30e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+005.52e-05Japanese Electric GridNaphthalene <t< td=""><td>LPG Production</td><td>Formaldehvde</td><td>secondary</td><td>1.15e+02</td><td>2.28e-03</td></t<>	LPG Production	Formaldehvde	secondary	1.15e+02	2.28e-03
Panel componentsTolueneprimary9.09e+011.80e-03BacklightEthanolprimary7.24e+011.44e-03Japanese Electric GridFormaldehydemodel/secondary7.02e+011.39e-03Natural Gas Prod.Acetaldehydesecondary3.87e+017.68e-04PWB Mfg.Formaldehydemodel/secondary3.50e+016.94e-04LPG ProductionPropionaldehydesecondary3.49e+016.93e-04US electric gridPropionaldehydemodel/secondary1.89e+013.75e-04Natural Gas Prod.Benzenesecondary1.85e+013.67e-04Japanese Electric GridAcroleinmodel/secondary1.62e+013.21e-04Japanese Electric GridBenzyl chloridemodel/secondary1.28e+012.55e-04Japanese Electric GridToluenemodel/secondary8.03e+001.59e-04Fuel Oil #4 Prod.Acetaldehydesecondary3.89e+007.10e-04Fuel Oil #6 Prod.Acetaldehydesecondary3.73e+007.30e+00Japanese Electric GridNaphthalenemodel/secondary3.73e+007.30e+00Japanese Electric GridNaphthalenemodel/secondary3.73e+007.30e+00Japanese Electric GridNaphthalenemodel/secondary3.73e+005.52e-05Japanese Electric GridCarbon disulfidemodel/secondary2.78e+005.52e-05Japanese Electric GridCarbon disulfidemodel/secondary2.59e+005.14e-05Japanese El	LPG Production	Ammonia	secondary	1.05e+02	2.08e-03
BacklightEthanolprimary7.24e+011.44e-03Japanese Electric GridFormaldehydemodel/secondary7.02e+011.39e-03Natural Gas Prod.Acetaldehydesecondary3.87e+017.68e-04PWB Mfg.Formaldehydemodel/secondary3.50e+016.94e-04LPG ProductionPropionaldehydesecondary3.49e+016.93e-04US electric gridPropionaldehydemodel/secondary1.85e+013.75e-04Natural Gas Prod.Benzenesecondary1.85e+013.67e-04Japanese Electric GridAcroleinmodel/secondary1.62e+013.21e-04Japanese Electric GridBenzyl chloridemodel/secondary1.28e+012.55e-04Japanese Electric GridToluenemodel/secondary8.03e+001.59e-04Fuel Oil #4 Prod.Acetaldehydesecondary3.89e+001.71e-05Japanese Electric GridNaphthalenesecondary3.89e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.89e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+007.39e+00Japanese Electric GridCarbon disulfidemodel/secondary2.78e+005.52e-05Monitor/moduleAcetoneprimary2.59e+005.14e-05Japanese Electric GridMethyl ethyl ketonemodel/secondary2.21e+004.38e-05	Panel components	Toluene	primary	9.09e+01	1.80e-03
Japanese Electric GridFormaldehydemodel/secondary7.02e+011.39e-03Natural Gas Prod.Acetaldehydesecondary3.87e+017.68e-04PWB Mfg.Formaldehydemodel/secondary3.50e+016.94e-04LPG ProductionPropionaldehydesecondary3.49e+016.93e-04US electric gridPropionaldehydemodel/secondary1.89e+013.75e-04Natural Gas Prod.Benzenesecondary1.85e+013.67e-04Japanese Electric GridAcroleinmodel/secondary1.62e+013.21e-04Japanese Electric GridBenzyl chloridemodel/secondary1.28e+012.55e-04Japanese Electric GridToluenemodel/secondary8.03e+001.59e-04Fuel Oil #4 Prod.Acetaldehydesecondary3.89e+001.27e-04LPG ProductionBenzenesecondary3.89e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+007.39e-05Japanese Electric GridNaphthalenemodel/secondary2.78e+005.52e-05Japanese Electric GridCarbon disulfidemodel/secondary2.78e+005.52e-05Japanese Electric GridAcetoneprimary2.59e+005.14e-05Japanese Electric GridMethyl ethyl ethyl ethoremodel/secondary2.21e+004.38e-05	Backlight	Ethanol	primary	7.24e+01	1.44e-03
Natural Gas Prod.Acetaldehydesecondary3.87e+017.68e-04PWB Mfg.Formaldehydemodel/secondary3.50e+016.94e-04LPG ProductionPropionaldehydesecondary3.49e+016.93e-04US electric gridPropionaldehydemodel/secondary1.89e+013.75e-04Natural Gas Prod.Benzenesecondary1.85e+013.67e-04Japanese Electric GridAceroleinmodel/secondary1.62e+013.21e-04Japanese Electric GridBenzyl chloridemodel/secondary1.28e+012.55e-04Japanese Electric GridToluenemodel/secondary1.28e+001.59e-04Fuel Oil #4 Prod.Acetaldehydesecondary7.30e+001.45e-04LPG ProductionBenzenesecondary3.89e+001.27e-04Fuel Oil #6 Prod.Acetaldehydesecondary3.89e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.89e+007.39e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+007.39e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+007.39e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+005.52e-05Monitor/moduleAcetoneprimary2.59e+005.14e-05Japanese Electric GridMethyl ethyl ketonemodel/secondary2.21e+004.38e-05	Japanese Electric Grid	Formaldehvde	model/secondary	7.02e+01	1.39e-03
PWB Mfg.Formaldehydemodel/secondary3.50e+016.94e-04LPG ProductionPropionaldehydesecondary3.49e+016.93e-04US electric gridPropionaldehydemodel/secondary1.89e+013.75e-04Natural Gas Prod.Benzenesecondary1.85e+013.67e-04Japanese Electric GridAcroleinmodel/secondary1.62e+013.21e-04Japanese Electric GridBenzyl chloridemodel/secondary1.28e+012.55e-04Japanese Electric GridToluenemodel/secondary1.28e+012.55e-04Japanese Electric GridToluenemodel/secondary8.03e+001.59e-04Fuel Oil #4 Prod.Acetaldehydesecondary7.30e+001.45e-04LPG ProductionBenzenesecondary3.89e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+007.39e-05Japanese Electric GridCarbon disulfidemodel/secondary2.78e+005.52e-05Monitor/moduleAcetoneprimary2.59e+005.14e-05Japanese Electric GridMethyl ethyl ketonemodel/secondary2.21e+004.38e-05	Natural Gas Prod.	Acetaldehvde	secondary	3.87e+01	7.68e-04
LPG ProductionPropionaldehydesecondary3.49e+016.93e-04US electric gridPropionaldehydemodel/secondary1.89e+013.75e-04Natural Gas Prod.Benzenesecondary1.85e+013.67e-04Japanese Electric GridAcroleinmodel/secondary1.62e+013.21e-04Japanese Electric GridBenzyl chloridemodel/secondary1.28e+012.55e-04Japanese Electric GridToluenemodel/secondary1.28e+012.55e-04Japanese Electric GridToluenemodel/secondary8.03e+001.59e-04Fuel Oil #4 Prod.Acetaldehydesecondary7.30e+001.45e-04LPG ProductionBenzenesecondary6.39e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+007.39e-05Japanese Electric GridCarbon disulfidemodel/secondary2.78e+005.52e-05Monitor/moduleAcetoneprimary2.59e+005.14e-05Japanese Electric GridMethyl ethyl ketonemodel/secondary2.21e+004.38e-05	PWB Mfg.	Formaldehvde	model/secondary	3.50e+01	6.94e-04
US electric gridPropionaldehydemodel/secondary1.89e+013.75e-04Natural Gas Prod.Benzenesecondary1.85e+013.67e-04Japanese Electric GridAcroleinmodel/secondary1.62e+013.21e-04Japanese Electric GridBenzyl chloridemodel/secondary1.28e+012.55e-04Japanese Electric GridToluenemodel/secondary8.03e+001.59e-04Fuel Oil #4 Prod.Acetaldehydesecondary7.30e+001.45e-04LPG ProductionBenzenesecondary3.89e+007.71e-05Fuel Oil #6 Prod.Acetaldehydesecondary3.73e+007.39e-05Japanese Electric GridNaphthalenemodel/secondary2.78e+005.52e-05Monitor/moduleAcetoneprimary2.59e+005.14e-05Japanese Electric GridMethyl ethyl ketonemodel/secondary2.21e+004.38e-05	LPG Production	Propionaldehyde	secondary	3.49e+01	6.93e-04
Natural Gas Prod.Benzenesecondary1.85e+013.67e-04Japanese Electric GridAcroleinmodel/secondary1.62e+013.21e-04Japanese Electric GridBenzyl chloridemodel/secondary1.28e+012.55e-04Japanese Electric GridToluenemodel/secondary8.03e+001.59e-04Fuel Oil #4 Prod.Acetaldehydesecondary7.30e+001.45e-04LPG ProductionBenzenesecondary6.39e+001.27e-04Fuel Oil #6 Prod.Acetaldehydesecondary3.89e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+007.39e-05Japanese Electric GridCarbon disulfidemodel/secondary2.78e+005.52e-05Monitor/moduleAcetoneprimary2.59e+005.14e-05Japanese Electric GridMethyl ethyl ketonemodel/secondary2.21e+004.38e-05	US electric grid	Propionaldehyde	model/secondary	1.89e+01	3.75e-04
Japanese Electric GridAcroleinmodel/secondary1.62e+013.21e-04Japanese Electric GridBenzyl chloridemodel/secondary1.28e+012.55e-04Japanese Electric GridToluenemodel/secondary8.03e+001.59e-04Fuel Oil #4 Prod.Acetaldehydesecondary7.30e+001.45e-04LPG ProductionBenzenesecondary6.39e+001.27e-04Fuel Oil #6 Prod.Acetaldehydesecondary3.89e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+007.39e-05Japanese Electric GridCarbon disulfidemodel/secondary2.78e+005.52e-05Monitor/moduleAcetoneprimary2.59e+005.14e-05Japanese Electric GridMethyl ethyl ketonemodel/secondary2.21e+004.38e-05	Natural Gas Prod.	Benzene	secondary	1.85e+01	3.67e-04
Japanese Electric GridBenzyl chloridemodel/secondary1.28e+012.55e-04Japanese Electric GridToluenemodel/secondary8.03e+001.59e-04Fuel Oil #4 Prod.Acetaldehydesecondary7.30e+001.45e-04LPG ProductionBenzenesecondary6.39e+001.27e-04Fuel Oil #6 Prod.Acetaldehydesecondary3.89e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+007.39e-05Japanese Electric GridCarbon disulfidemodel/secondary2.78e+005.52e-05Monitor/moduleAcetoneprimary2.59e+005.14e-05Japanese Electric GridMethyl ethyl ketonemodel/secondary2.21e+004.38e-05	Japanese Electric Grid	Acrolein	model/secondary	1.62e+01	3.21e-04
Japanese Electric GridToluenemodel/secondary8.03e+001.59e-04Fuel Oil #4 Prod.Acetaldehydesecondary7.30e+001.45e-04LPG ProductionBenzenesecondary6.39e+001.27e-04Fuel Oil #6 Prod.Acetaldehydesecondary3.89e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+007.39e-05Japanese Electric GridCarbon disulfidemodel/secondary2.78e+005.52e-05Monitor/moduleAcetoneprimary2.59e+005.14e-05Japanese Electric GridMethyl ethyl ketonemodel/secondary2.21e+004.38e-05	Japanese Electric Grid	Benzyl chloride	model/secondary	1.28e+01	2.55e-04
Fuel Oil #4 Prod.Acetaldehydesecondary7.30e+001.45e-04LPG ProductionBenzenesecondary6.39e+001.27e-04Fuel Oil #6 Prod.Acetaldehydesecondary3.89e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+007.39e-05Japanese Electric GridCarbon disulfidemodel/secondary2.78e+005.52e-05Monitor/moduleAcetoneprimary2.59e+005.14e-05Japanese Electric GridMethyl ethyl ketonemodel/secondary2.21e+004.38e-05	Japanese Electric Grid	Toluene	model/secondary	8.03e+00	1.59e-04
LPG ProductionBenzenesecondary6.39e+001.27e-04Fuel Oil #6 Prod.Acetaldehydesecondary3.89e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+007.39e-05Japanese Electric GridCarbon disulfidemodel/secondary2.78e+005.52e-05Monitor/moduleAcetoneprimary2.59e+005.14e-05Japanese Electric GridMethyl ethyl ketonemodel/secondary2.21e+004.38e-05	Fuel Oil #4 Prod.	Acetaldehvde	secondary	7.30e+00	1.45e-04
Fuel Oil #6 Prod.Acetaldehydesecondary3.89e+007.71e-05Japanese Electric GridNaphthalenemodel/secondary3.73e+007.39e-05Japanese Electric GridCarbon disulfidemodel/secondary2.78e+005.52e-05Monitor/moduleAcetoneprimary2.59e+005.14e-05Japanese Electric GridMethyl ethyl ketonemodel/secondary2.21e+004.38e-05	LPG Production	Benzene	secondary	6.39e+00	1.27e-04
Japanese Electric GridNaphthalenemodel/secondary3.73e+007.39e-05Japanese Electric GridCarbon disulfidemodel/secondary2.78e+005.52e-05Monitor/moduleAcetoneprimary2.59e+005.14e-05Japanese Electric GridMethyl ethyl ketonemodel/secondary2.21e+004.38e-05	Fuel Oil #6 Prod.	Acetaldehyde	secondary	3.89e+00	7.71e-05
Japanese Electric GridCarbon disulfidemodel/secondary2.78e+005.52e-05Monitor/moduleAcetoneprimary2.59e+005.14e-05Japanese Electric GridMethyl ethyl ketonemodel/secondary2.21e+004.38e-05	Japanese Electric Grid	Naphthalene	model/secondary	3.73e+00	7.39e-05
Monitor/moduleAcetoneprimary2.59e+005.14e-05Japanese Electric GridMethyl ethyl ketonemodel/secondary2.21e+004.38e-05	Japanese Electric Grid	Carbon disulfide	model/secondary	2.78e+00	5.52e-05
Japanese Electric Grid Methyl ethyl ketone model/secondary 2.21e+00 4.38e-05	Monitor/module	Acetone	primary	2.59e+00	5.14e-05
1	Japanese Electric Grid	Methyl ethyl ketone	model/secondary	2.21e+00	4.38e-05

Table M-36. LCD LCIA Results for the Aesthetics Impact Category

Process Group	Material	LCI Data Type	Odor (m3)	% of Total
Fuel Oil #2 Prod.	Acetaldehyde	secondary	2.07e+00	4.11e-05
Japanese Electric Grid	Isophorone	model/secondary	2.03e+00	4.03e-05
Natural Gas Prod.	Propionaldehyde	secondary	1.99e+00	3.95e-05
Japanese Electric Grid	Phenol	model/secondary	1.58e+00	3.13e-05
Japanese Electric Grid	Styrene	model/secondary	1.42e+00	2.81e-05
Fuel Oil #4 Prod.	Formaldehyde	secondary	1.41e+00	2.80e-05
LPG Production	Acrolein	secondary	1.35e+00	2.68e-05
Panel components	Methyl ethyl ketone	primary	1.08e+00	2.14e-05
LPG Production	Benzyl chloride	secondary	1.07e+00	2.13e-05
Fuel Oil #4 Prod.	Ammonia	secondary	9.19e-01	1.82e-05
Fuel Oil #6 Prod.	Formaldehyde	secondary	8.33e-01	1.65e-05
Natural Gas Prod.	Formaldehyde	secondary	7.80e-01	1.55e-05
US electric grid	Acrolein	model/secondary	7.31e-01	1.45e-05
US electric grid	Benzyl chloride	model/secondary	5.80e-01	1.15e-05
Fuel Oil #6 Prod.	Ammonia	secondary	4.00e-01	7.93e-06
Japanese Electric Grid	Methyl methacrylate	model/secondary	3.85e-01	7.64e-06
Fuel Oil #4 Prod.	Propionaldehyde	secondary	3.75e-01	7.44e-06
Fuel Oil #2 Prod.	Formaldehyde	secondary	3.67e-01	7.29e-06
Japanese Electric Grid	Methyl hydrazine	model/secondary	3.45e-01	6.84e-06
US electric grid	Formaldehyde	model/secondary	3.35e-01	6.65e-06
Fuel Oil #2 Prod.	Ammonia	secondary	3.00e-01	5.95e-06
LPG Production	Toluene	secondary	2.98e-01	5.91e-06
Japanese Electric Grid	2-Chloroacetophenone	model/secondary	2.70e-01	5.35e-06
Japanese Electric Grid	Xylene (mixed isomers)	model/secondary	2.34e-01	4.63e-06
LPG Production	Carbon disulfide	secondary	2.32e-01	4.61e-06
Fuel Oil #6 Prod.	Propionaldehyde	secondary	2.00e-01	3.97e-06
LPG Production	Methyl ethyl ketone	secondary	1.85e-01	3.66e-06
LPG Production	Isophorone	secondary	1.70e-01	3.36e-06
Natural Gas Prod.	o-xylene	secondary	1.40e-01	2.78e-06
Natural Gas Prod.	Toluene	secondary	1.36e-01	2.70e-06
LPG Production	Phenol	secondary	1.32e-01	2.62e-06
US electric grid	Carbon disulfide	model/secondary	1.26e-01	2.49e-06
LPG Production	Styrene	secondary	1.18e-01	2.35e-06
Fuel Oil #2 Prod.	Propionaldehyde	secondary	1.07e-01	2.11e-06
US electric grid	Methyl ethyl ketone	model/secondary	9.98e-02	1.98e-06
US electric grid	Isophorone	model/secondary	9.18e-02	1.82e-06
Japanese Electric Grid	Chlorobenzene	model/secondary	8.48e-02	1.68e-06
LPG Production	Naphthalene	secondary	8.39e-02	1.66e-06
Japanese Electric Grid	o-xylene	model/secondary	8.34e-02	1.65e-06
US electric grid	Toluene	model/secondary	8.22e-02	1.63e-06
Natural Gas Prod.	Acrolein	secondary	7.71e-02	1.53e-06
Japanese Electric Grid	Vinyl acetate	model/secondary	7.31e-02	1.45e-06
US electric grid	Phenol	model/secondary	7.14e-02	1.42e-06
US electric grid	Styrene	model/secondary	6.40e-02	1.27e-06
LPG Production	o-xylene	secondary	6.36e-02	1.26e-06
Natural Gas Prod.	Benzyl chloride	secondary	6.11e-02	1.21e-06
Panel components	Ethylacetate	primary	5.82e-02	1.15e-06
Fuel Oil #4 Prod.	Benzene	secondary	5.77e-02	1.14e-06
Japanese Electric Grid	Benzene	model/secondary	4.76e-02	9.43e-07
LPG Production	Cumene	secondary	4.26e-02	8.45e-07
Japanese Electric Grid	1,1,1-Trichloroethane	model/secondary	4.11e-02	8.15e-07

 Table M-36.
 LCD LCIA Results for the Aesthetics Impact Category

Process Group	Material	LCI Data Type	Odor (m3)	% of Total
Japanese Electric Grid	Acetophenone	model/secondary	3.85e-02	7.64e-07
LPG Production	Methyl methacrylate	secondary	3.22e-02	6.38e-07
LPG Production	Methyl hydrazine	secondary	2.88e-02	5.71e-07
Fuel Oil #6 Prod.	Benzene	secondary	2.59e-02	5.13e-07
US electric grid	Cumene	model/secondary	2.31e-02	4.57e-07
LPG Production	2-Chloroacetophenone	secondary	2.25e-02	4.47e-07
Japanese Electric Grid	Tetrachloroethylene	model/secondary	2.00e-02	3.96e-07
LPG Production	1,4-Dichlorobenzene	secondary	1.98e-02	3.92e-07
Fuel Oil #2 Prod.	Benzene	secondary	1.85e-02	3.66e-07
US electric grid	Naphthalene	model/secondary	1.77e-02	3.52e-07
US electric grid	Methyl methacrylate	model/secondary	1.74e-02	3.45e-07
US electric grid	Methyl hydrazine	model/secondary	1.56e-02	3.09e-07
Fuel Oil #4 Prod.	Acrolein	secondary	1.45e-02	2.88e-07
Natural Gas Prod.	Naphthalene	secondary	1.39e-02	2.76e-07
Natural Gas Prod.	Carbon disulfide	secondary	1.32e-02	2.63e-07
US electric grid	2-Chloroacetophenone	model/secondary	1.22e-02	2.42e-07
Fuel Oil #4 Prod.	Benzyl chloride	secondary	1.15e-02	2.28e-07
US electric grid	Xylene (mixed isomers)	model/secondary	1.06e-02	2.09e-07
Natural Gas Prod.	Methyl ethyl ketone	secondary	1.05e-02	2.09e-07
Natural Gas Prod.	Isophorone	secondary	9.67e-03	1.92e-07
Fuel Oil #6 Prod.	Acrolein	secondary	7.74e-03	1.54e-07
Natural Gas Prod.	Phenol	secondary	7.52e-03	1.49e-07
LPG Production	Chlorobenzene	secondary	7.08e-03	1.40e-07
Natural Gas Prod.	Styrene	secondary	6.74e-03	1.34e-07
Japanese Electric Grid	1,2-Dichloroethane	model/secondary	6.17e-03	1.22e-07
Fuel Oil #6 Prod.	Benzyl chloride	secondary	6.14e-03	1.22e-07
LPG Production	Vinyl acetate	secondary	6.11e-03	1.21e-07
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	5.85e-03	1.16e-07
Fuel Oil #2 Prod.	Acrolein	secondary	4.13e-03	8.18e-08
US electric grid	Chlorobenzene	model/secondary	3.83e-03	7.59e-08
US electric grid	Vinyl acetate	model/secondary	3.31e-03	6.56e-08
Fuel Oil #2 Prod.	Benzyl chloride	secondary	3.27e-03	6.49e-08
Natural Gas Prod.	Chlorine	secondary	3.25e-03	6.45e-08
LPG Production	Acetophenone	secondary	3.22e-03	6.38e-08
Fuel Oil #4 Prod.	Toluene	secondary	2.91e-03	5.77e-08
Fuel Oil #4 Prod.	Carbon disulfide	secondary	2.50e-03	4.95e-08
Natural Gas Prod.	Cumene	secondary	2.43e-03	4.82e-08
US electric grid	Benzene	model/secondary	2.10e-03	4.16e-08
Fuel Oil #4 Prod.	Methyl ethyl ketone	secondary	1.98e-03	3.93e-08
Natural Gas Prod.	Methyl methacrylate	secondary	1.83e-03	3.64e-08
Fuel Oil #4 Prod.	Isophorone	secondary	1.82e-03	3.61e-08
Japanese Electric Grid	Dichloromethane	model/secondary	1 75e-03	3 46e-08
US electric grid	Acetophenone	model/secondary	1 74e-03	3 45e-08
LPG Production	Tetrachloroethylene	secondary	1.67e-03	3 31e-08
Natural Gas Prod	Methyl hydrazine	secondary	1.64e-03	3 25e-08
Fuel Oil #6 Prod	Toluene	secondary	1 42e-03	2.83e-08
Fuel Oil #4 Prod	Phenol	secondary	1.42e-03	2.05e 08
Fuel Oil #6 Prod	Carbon disulfide	secondary	1.33e-03	2.64e-08
Natural Gas Prod	2-Chloroacetophenope	secondary	1 28e-03	2.540 08 2.55e-08
Fuel Oil #4 Prod	Styrene	secondary	1.200-03	2.550-08 2.52e-08
L PG Production	1.1.1-Trichloroethane	secondary	1 210 03	2.520 08
	1,1,1 IIICIIIOIOCUIAIIC	secondary	1.210-03	2.410-00

Table M-36. LCD LCIA Results for the Aesthetics Impact Category

Process Group	Material	LCI Data Type	Odor (m3)	% of Total
LPG Production	Chlorine	secondary	1.11e-03	2.21e-08
Fuel Oil #6 Prod.	Methyl ethyl ketone	secondary	1.06e-03	2.10e-08
Fuel Oil #6 Prod.	Isophorone	secondary	9.71e-04	1.93e-08
US electric grid	Tetrachloroethylene	model/secondary	9.02e-04	1.79e-08
Fuel Oil #2 Prod.	Toluene	secondary	8.82e-04	1.75e-08
Fuel Oil #4 Prod.	Naphthalene	secondary	7.79e-04	1.54e-08
Fuel Oil #6 Prod.	Phenol	secondary	7.56e-04	1.50e-08
Fuel Oil #2 Prod.	Carbon disulfide	secondary	7.09e-04	1.41e-08
US electric grid	1,1,1-Trichloroethane	model/secondary	7.02e-04	1.39e-08
Fuel Oil #6 Prod.	Styrene	secondary	6.77e-04	1.34e-08
Fuel Oil #4 Prod.	o-xylene	secondary	5.99e-04	1.19e-08
Fuel Oil #2 Prod.	Methyl ethyl ketone	secondary	5.63e-04	1.12e-08
Fuel Oil #2 Prod.	Isophorone	secondary	5.18e-04	1.03e-08
LPG Production	1,2-Dichloroethane	secondary	5.15e-04	1.02e-08
Fuel Oil #4 Prod.	Cumene	secondary	4.58e-04	9.08e-09
Natural Gas Prod.	Chlorobenzene	secondary	4.03e-04	8.00e-09
Fuel Oil #2 Prod.	Phenol	secondary	4.03e-04	7.99e-09
Fuel Oil #2 Prod.	Styrene	secondary	3.61e-04	7.16e-09
Fuel Oil #6 Prod.	Naphthalene	secondary	3.61e-04	7.16e-09
Japanese Electric Grid	Chloroform	model/secondary	3.50e-04	6.94e-09
Natural Gas Prod.	Vinyl acetate	secondary	3.48e-04	6.91e-09
Fuel Oil #4 Prod.	Methyl methacrylate	secondary	3.46e-04	6.85e-09
Fuel Oil #4 Prod.	Methyl hydrazine	secondary	3.09e-04	6.13e-09
Fuel Oil #6 Prod.	o-xylene	secondary	2.82e-04	5.59e-09
US electric grid	1,2-Dichloroethane	model/secondary	2.78e-04	5.52e-09
Fuel Oil #2 Prod.	Naphthalene	secondary	2.45e-04	4.85e-09
Fuel Oil #6 Prod.	Cumene	secondary	2.44e-04	4.84e-09
Fuel Oil #4 Prod.	2-Chloroacetophenone	secondary	2.42e-04	4.80e-09
Fuel Oil #2 Prod.	o-xylene	secondary	1.86e-04	3.69e-09
Fuel Oil #6 Prod.	Methyl methacrylate	secondary	1.84e-04	3.65e-09
Natural Gas Prod.	Acetophenone	secondary	1.83e-04	3.64e-09
Fuel Oil #4 Prod.	1,4-Dichlorobenzene	secondary	1.79e-04	3.54e-09
Fuel Oil #6 Prod.	Methyl hydrazine	secondary	1.65e-04	3.27e-09
LPG Production	Dichloromethane	secondary	1.46e-04	2.89e-09
US electric grid	o-xylene	model/secondary	1.42e-04	2.82e-09
Fuel Oil #2 Prod.	Cumene	secondary	1.30e-04	2.58e-09
Fuel Oil #6 Prod.	2-Chloroacetophenone	secondary	1.29e-04	2.56e-09
Fuel Oil #2 Prod.	Methyl methacrylate	secondary	9.82e-05	1.95e-09
Natural Gas Prod.	Tetrachloroethylene	secondary	9.50e-05	1.88e-09
Fuel Oil #2 Prod.	Methyl hydrazine	secondary	8.79e-05	1.74e-09
Fuel Oil #6 Prod.	1,4-Dichlorobenzene	secondary	8.02e-05	1.59e-09
US electric grid	Dichloromethane	model/secondary	7.89e-05	1.56e-09
Fuel Oil #4 Prod.	Chlorobenzene	secondary	7.60e-05	1.51e-09
Natural Gas Prod.	1,1,1-Trichloroethane	secondary	6.92e-05	1.37e-09
Fuel Oil #2 Prod.	2-Chloroacetophenone	secondary	6.87e-05	1.36e-09
Fuel Oil #4 Prod.	Vinyl acetate	secondary	6.57e-05	1.30e-09
Fuel Oil #2 Prod.	1,4-Dichlorobenzene	secondary	5.72e-05	1.13e-09
Fuel Oil #6 Prod.	Chlorobenzene	secondary	4.05e-05	8.04e-10
Fuel Oil #6 Prod.	Vinyl acetate	secondary	3.50e-05	6.94e-10
Fuel Oil #4 Prod.	Acetophenone	secondary	3.46e-05	6.85e-10
Natural Gas Prod.	1,2-Dichloroethane	secondary	2.93e-05	5.82e-10

 Table M-36.
 LCD LCIA Results for the Aesthetics Impact Category

Process Group	Material	LCI Data Type	Odor (m3)	% of Total
LPG Production	Chloroform	secondary	2.92e-05	5.79e-10
Fuel Oil #2 Prod.	Chlorobenzene	secondary	2.16e-05	4.28e-10
Fuel Oil #2 Prod.	Vinyl acetate	secondary	1.87e-05	3.70e-10
Fuel Oil #6 Prod.	Acetophenone	secondary	1.84e-05	3.65e-10
Fuel Oil #4 Prod.	Tetrachloroethylene	secondary	1.79e-05	3.55e-10
US electric grid	Chloroform	model/secondary	1.58e-05	3.13e-10
Fuel Oil #4 Prod.	1,1,1-Trichloroethane	secondary	1.30e-05	2.59e-10
Fuel Oil #4 Prod.	Chlorine	secondary	9.86e-06	1.95e-10
Fuel Oil #2 Prod.	Acetophenone	secondary	9.82e-06	1.95e-10
Fuel Oil #6 Prod.	Tetrachloroethylene	secondary	9.54e-06	1.89e-10
Natural Gas Prod.	Dichloromethane	secondary	8.31e-06	1.65e-10
Fuel Oil #6 Prod.	1,1,1-Trichloroethane	secondary	6.95e-06	1.38e-10
Fuel Oil #4 Prod.	1,2-Dichloroethane	secondary	5.53e-06	1.10e-10
Fuel Oil #2 Prod.	Tetrachloroethylene	secondary	5.09e-06	1.01e-10
Fuel Oil #6 Prod.	Chlorine	secondary	4.39e-06	8.71e-11
Fuel Oil #2 Prod.	1,1,1-Trichloroethane	secondary	3.71e-06	7.35e-11
Fuel Oil #2 Prod.	Chlorine	secondary	3.17e-06	6.29e-11
Fuel Oil #6 Prod.	1,2-Dichloroethane	secondary	2.95e-06	5.85e-11
Natural Gas Prod.	Chloroform	secondary	1.66e-06	3.30e-11
Fuel Oil #2 Prod.	1,2-Dichloroethane	secondary	1.57e-06	3.12e-11
Fuel Oil #4 Prod.	Dichloromethane	secondary	1.57e-06	3.11e-11
Fuel Oil #6 Prod.	Dichloromethane	secondary	8.35e-07	1.66e-11
Fuel Oil #2 Prod.	Dichloromethane	secondary	4.45e-07	8.82e-12
Fuel Oil #4 Prod.	Chloroform	secondary	3.14e-07	6.22e-12
Fuel Oil #6 Prod.	Chloroform	secondary	1.67e-07	3.32e-12
Fuel Oil #2 Prod.	Chloroform	secondary	8.91e-08	1.77e-12
Total Manufacturing			4.93e+06	9.77e+01
Use, Maintenance and Repair Life-cy	vcle Stage			
US electric grid	Acetaldehvde	model/secondary	7.08e+04	1.40e+00
US electric grid	Propionaldehyde	model/secondary	3.64e+03	7.22e-02
US electric grid	Acrolein	model/secondary	1.41e+02	2.80e-03
US electric grid	Benzyl chloride	model/secondary	1.12e+02	2.22e-03
US electric grid	Formaldehyde	model/secondary	6.46e+01	1.28e-03
US electric grid	Carbon disulfide	model/secondary	2.42e+01	4.80e-04
US electric grid	Methyl ethyl ketone	model/secondary	1.92e+01	3.82e-04
US electric grid	Isophorone	model/secondary	1.77e+01	3.51e-04
US electric grid	Toluene	model/secondary	1.58e+01	3.14e-04
US electric grid	Phenol	model/secondary	1.38e+01	2.73e-04
US electric grid	Styrene	model/secondary	1.23e+01	2.45e-04
US electric grid	Cumene	model/secondary	4.45e+00	8.81e-05
US electric grid	Naphthalene	model/secondary	3.42e+00	6.78e-05
US electric grid	Methyl methacrylate	model/secondary	3.35e+00	6.65e-05
US electric grid	Methyl hydrazine	model/secondary	3.00e+00	5.95e-05
US electric grid	2-Chloroacetophenone	model/secondary	2.35e+00	4.66e-05
US electric grid	Xvlene (mixed isomers)	model/secondary	2.03e+00	4.04e-05
US electric grid	Chlorobenzene	model/secondary	7.38e-01	1.46e-05
US electric grid	Vinyl acetate	model/secondary	6.37e-01	1.26e-05
US electric grid	Benzene	model/secondary	4.04e-01	8.02e-06
US electric grid	Acetophenone	model/secondary	3.35e-01	6.65e-06
US electric grid	Tetrachloroethylene	model/secondary	1.74e-01	3.45e-06
US electric grid	1.1.1-Trichloroethane	model/secondary	1.35e-01	2.68e-06
	· · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·		

 Table M-36.
 LCD LCIA Results for the Aesthetics Impact Category

Process Group	Material	LCI Data Type	Odor (m3)	% of Total
US electric grid	1,2-Dichloroethane	model/secondary	5.37e-02	1.06e-06
US electric grid	o-xylene	model/secondary	2.74e-02	5.43e-07
US electric grid	Dichloromethane	model/secondary	1.52e-02	3.01e-07
US electric grid	Chloroform	model/secondary	3.05e-03	6.04e-08
Total Use, Maintenance and Repair			7.49e+04	1.49e+00
End-of-life Life-cycle Stage				
LCD landfilling	Hydrogen sulfide	primary	1 29e+02	2.56e-03
LPG Production	Hydrogen sulfide	secondary	2.81e+01	5.58e-04
US electric grid	Acetaldehvde	model/secondary	1.34e+01	2.66e-04
US electric grid	Propionaldehyde	model/secondary	6.91e-01	1.37e-05
LPG Production	Acetaldehyde	secondary	5 58e-02	1 11e-06
US electric grid	Acrolein	model/secondary	2.68e-02	5 31e-07
US electric grid	Benzyl chloride	model/secondary	2.12e-02	4 21e-07
LCD landfilling	Toluene	nrimary	1.92e-02	3.81e-07
US electric orid	Formaldehyde	model/secondary	1.32e 02	2 43e-07
LCD landfilling	Xylene (mixed isomers)	nrimary	9.80e-03	1 94e-07
LPG Production	Formaldehyde	secondary	9.43e-03	1.9 Te -07
LPG Production	Ammonia	secondary	8 59e-03	1.0/e 07
US electric grid	Carbon disulfide	model/secondary	4 60e-03	9.12e-08
US electric grid	Methyl ethyl ketone	model/secondary	3.65e-03	7.24e-08
US electric grid	Isophorone	model/secondary	3.36e-03	6 66e-08
US electric grid	Toluene	model/secondary	3.01e-03	5.96e-08
LPG Production	Propionaldehyde	secondary	2.87e-03	5.50e 08
US electric grid	Phenol	model/secondary	2.67c-03	5.18e-08
US electric grid	Styrene	model/secondary	2.010-03 2.34e-03	4 64e-08
I CD landfilling	Benzene	nrimary	1.51e-03	3.00e-08
US electric grid	Cumene	model/secondary	8.43e-04	1.67e-08
US electric grid	Nanhthalene	model/secondary	6.49e-04	1.07e 08
US electric grid	Methyl methacrylate	model/secondary	6.37e-04	1.25e-08
US electric grid	Methyl hydrazine	model/secondary	5 70e-04	1.13e-08
LPG Production	Benzene	secondary	5.76e 01	1.04e-08
US electric grid	2-Chloroacetophenone	model/secondary	4 46e-04	8 84e-09
US electric grid	Xylene (mixed isomers)	model/secondary	3 86e-04	7.66e-09
US electric grid	Chlorobenzene	model/secondary	1 40e-04	2 78e-09
US electric grid	Vinyl acetate	model/secondary	1 21e-04	2.40e-09
LPG Production	Acrolein	secondary	1 11e-04	2.20e-09
LPG Production	Benzyl chloride	secondary	8 81e-05	1 75e-09
LCD landfilling	Trichloroethylene	nrimary	8 47e-05	1.68e-09
US electric grid	Benzene	model/secondary	7 67e-05	1.50e-09
LCD incineration	Trichloroethylene	secondary	6 57e-05	1 30e-09
US electric orid	Acetophenone	model/secondary	6 37e-05	1.26e-09
LCD landfilling	Tetrachloroethylene	nrimary	3.98e-05	7 90e-10
US electric orid	Tetrachloroethylene	model/secondary	3 30e-05	6 54e-10
US electric grid	1 1 1-Trichloroethane	model/secondary	2 57e-05	5.09e-10
LPG Production	Toluene	secondary	2.67e-05	4 85e-10
LPG Production	Carbon disulfide	secondary	1.91e-05	3 78e-10
LPG Production	Methyl ethyl ketone	secondary	1.57e-05	3.00e-10
LPG Production	Isophorone	secondary	1.39e-05	2.76e-10
LCD landfilling	1 2-Dichloroethane	primary	1.37e-05	2.700-10 2.62e-10
LPG Production	Phenol	secondary	1.08e-05	2.020 10 2.15e-10
LI G I louuvuoli	1 Hendi	secondary	1.000-05	2.130-10

 Table M-36.
 LCD LCIA Results for the Aesthetics Impact Category

Process Group	Material	LCI Data Type	Odor (m3)	% of Total
US electric grid	1,2-Dichloroethane	model/secondary	1.02e-05	2.02e-10
LPG Production	Styrene	secondary	9.71e-06	1.93e-10
LPG Production	Naphthalene	secondary	6.89e-06	1.37e-10
LPG Production	o-xylene	secondary	5.22e-06	1.04e-10
US electric grid	o-xylene	model/secondary	5.19e-06	1.03e-10
LPG Production	Cumene	secondary	3.50e-06	6.94e-11
US electric grid	Dichloromethane	model/secondary	2.88e-06	5.72e-11
LPG Production	Methyl methacrylate	secondary	2.64e-06	5.24e-11
LPG Production	Methyl hydrazine	secondary	2.36e-06	4.69e-11
LPG Production	2-Chloroacetophenone	secondary	1.85e-06	3.67e-11
LPG Production	1,4-Dichlorobenzene	secondary	1.62e-06	3.22e-11
LCD landfilling	Dichloromethane	primary	8.25e-07	1.64e-11
LPG Production	Chlorobenzene	secondary	5.81e-07	1.15e-11
US electric grid	Chloroform	model/secondary	5.78e-07	1.15e-11
LCD landfilling	Chloroform	primary	5.08e-07	1.01e-11
LPG Production	Vinyl acetate	secondary	5.02e-07	9.95e-12
LCD landfilling	Carbon tetrachloride	primary	3.74e-07	7.41e-12
LCD incineration	Carbon tetrachloride	secondary	2.90e-07	5.75e-12
LPG Production	Acetophenone	secondary	2.64e-07	5.24e-12
LPG Production	Tetrachloroethylene	secondary	1.37e-07	2.71e-12
LPG Production	1,1,1-Trichloroethane	secondary	9.97e-08	1.98e-12
LPG Production	Chlorine	secondary	9.13e-08	1.81e-12
LPG Production	1,2-Dichloroethane	secondary	4.23e-08	8.38e-13
LPG Production	Dichloromethane	secondary	1.20e-08	2.37e-13
LPG Production	Chloroform	secondary	2.40e-09	4.75e-14
Natural Gas Prod.	Chloroform	secondary	-3.50e-07	-6.93e-12
Fuel Oil #4 Prod.	Chloroform	secondary	-1.45e-06	-2.87e-11
Natural Gas Prod.	Dichloromethane	secondary	-1.74e-06	-3.46e-11
Natural Gas Prod.	1,2-Dichloroethane	secondary	-6.16e-06	-1.22e-10
Fuel Oil #4 Prod.	Dichloromethane	secondary	-7.22e-06	-1.43e-10
Natural Gas Prod.	1,1,1-Trichloroethane	secondary	-1.45e-05	-2.88e-10
Natural Gas Prod.	Tetrachloroethylene	secondary	-2.00e-05	-3.96e-10
Fuel Oil #4 Prod.	1,2-Dichloroethane	secondary	-2.55e-05	-5.05e-10
Natural Gas Prod.	Acetophenone	secondary	-3.85e-05	-7.64e-10
Fuel Oil #4 Prod.	Chlorine	secondary	-4.54e-05	-9.01e-10
LCD incineration	Chlorine	secondary	-5.47e-05	-1.08e-09
Fuel Oil #4 Prod.	1,1,1-Trichloroethane	secondary	-6.01e-05	-1.19e-09
Natural Gas Prod.	Vinyl acetate	secondary	-7.32e-05	-1.45e-09
Fuel Oil #4 Prod.	Tetrachloroethylene	secondary	-8.25e-05	-1.64e-09
Natural Gas Prod.	Chlorobenzene	secondary	-8.47e-05	-1.68e-09
LCD incineration	Chloroform	secondary	-8.73e-05	-1.73e-09
Fuel Oil #4 Prod.	Acetophenone	secondary	-1.59e-04	-3.16e-09
Natural Gas Prod.	2-Chloroacetophenone	secondary	-2.70e-04	-5.35e-09
Fuel Oil #4 Prod.	Vinyl acetate	secondary	-3.03e-04	-6.00e-09
Natural Gas Prod.	Methyl hydrazine	secondary	-3.45e-04	-6.83e-09
Fuel Oil #4 Prod.	Chlorobenzene	secondary	-3.50e-04	-6.95e-09
Natural Gas Prod.	Methyl methacrylate	secondary	-3.85e-04	-7.64e-09
LCD incineration	Dichloromethane	secondary	-4.37e-04	-8.66e-09
Natural Gas Prod.	Cumene	secondary	-5.10e-04	-1.01e-08
Natural Gas Prod.	Chlorine	secondary	-6.83e-04	-1.35e-08
Fuel Oil #4 Prod.	1,4-Dichlorobenzene	secondary	-8.23e-04	-1.63e-08

 Table M-36.
 LCD LCIA Results for the Aesthetics Impact Category

Process Group	Material	LCI Data Type	Odor (m3)	% of Total
Fuel Oil #4 Prod.	2-Chloroacetophenone	secondary	-1.11e-03	-2.21e-08
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	-1.23e-03	-2.44e-08
Natural Gas Prod.	Styrene	secondary	-1.42e-03	-2.81e-08
Fuel Oil #4 Prod.	Methyl hydrazine	secondary	-1.43e-03	-2.83e-08
LCD incineration	1,2-Dichloroethane	secondary	-1.53e-03	-3.04e-08
Natural Gas Prod.	Phenol	secondary	-1.58e-03	-3.13e-08
Fuel Oil #4 Prod.	Methyl methacrylate	secondary	-1.59e-03	-3.16e-08
Natural Gas Prod.	Isophorone	secondary	-2.03e-03	-4.03e-08
Fuel Oil #4 Prod.	Cumene	secondary	-2.11e-03	-4.18e-08
Natural Gas Prod.	Methyl ethyl ketone	secondary	-2.21e-03	-4.38e-08
Fuel Oil #4 Prod.	o-xylene	secondary	-2.76e-03	-5.47e-08
Natural Gas Prod.	Carbon disulfide	secondary	-2.78e-03	-5.52e-08
Natural Gas Prod.	Naphthalene	secondary	-2.92e-03	-5.80e-08
Fuel Oil #4 Prod.	Naphthalene	secondary	-3.59e-03	-7.11e-08
LCD incineration	1,1,1-Trichloroethane	secondary	-3.64e-03	-7.23e-08
LCD incineration	Tetrachloroethylene	secondary	-4.97e-03	-9.86e-08
Fuel Oil #4 Prod.	Styrene	secondary	-5.86e-03	-1.16e-07
Fuel Oil #4 Prod.	Phenol	secondary	-6.53e-03	-1.30e-07
Fuel Oil #4 Prod.	Isophorone	secondary	-8.40e-03	-1.67e-07
Fuel Oil #4 Prod.	Methyl ethyl ketone	secondary	-9.13e-03	-1.81e-07
LCD incineration	Acetophenone	secondary	-9.66e-03	-1.91e-07
Fuel Oil #4 Prod.	Carbon disulfide	secondary	-1.15e-02	-2.28e-07
Natural Gas Prod.	Benzyl chloride	secondary	-1.28e-02	-2.55e-07
Fuel Oil #4 Prod.	Toluene	secondary	-1.34e-02	-2.66e-07
Natural Gas Prod.	Acrolein	secondary	-1.62e-02	-3.21e-07
LCD incineration	Vinyl acetate	secondary	-1.83e-02	-3.64e-07
LCD incineration	Chlorobenzene	secondary	-2.12e-02	-4.21e-07
Natural Gas Prod.	Toluene	secondary	-2.86e-02	-5.66e-07
Natural Gas Prod.	o-xylene	secondary	-2.94e-02	-5.84e-07
Fuel Oil #4 Prod.	Benzyl chloride	secondary	-5.31e-02	-1.05e-06
LCD incineration	Xylene (mixed isomers)	secondary	-5.48e-02	-1.09e-06
Fuel Oil #4 Prod.	Acrolein	secondary	-6.69e-02	-1.33e-06
LCD incineration	Methyl hydrazine	secondary	-8.64e-02	-1.71e-06
LCD incineration	Naphthalene	secondary	-8.91e-02	-1.77e-06
LCD incineration	Methyl methacrylate	secondary	-9.66e-02	-1.91e-06
LCD incineration	Cumene	secondary	-1.28e-01	-2.54e-06
Natural Gas Prod.	Formaldehyde	secondary	-1.64e-01	-3.25e-06
Fuel Oil #4 Prod.	Benzene	secondary	-2.66e-01	-5.27e-06
LCD incineration	Benzene	secondary	-3.23e-01	-6.40e-06
LCD incineration	Styrene	secondary	-3.55e-01	-7.04e-06
LCD incineration	Phenol	secondary	-3.96e-01	-7.86e-06
Natural Gas Prod.	Propionaldehyde	secondary	-4.18e-01	-8.29e-06
LCD incineration	Toluene	secondary	-4.73e-01	-9.38e-06
LCD incineration	Isophorone	secondary	-5.09e-01	-1.01e-05
LCD incineration	Methyl ethyl ketone	secondary	-5.54e-01	-1.10e-05
LCD incineration	Carbon disulfide	secondary	-6.97e-01	-1.38e-05
Fuel Oil #4 Prod.	Propionaldehyde	secondary	-1.73e+00	-3.43e-05
LCD incineration	Benzyl chloride	secondary	-3.22e+00	-6.38e-05
Natural Gas Prod.	Benzene	secondary	-3.89e+00	-7.72e-05
LCD incineration	Acrolein	secondary	-4.06e+00	-8.05e-05
Fuel Oil #4 Prod.	Ammonia	secondary	-4.24e+00	-8.40e-05
		-		

 Table M-36.
 LCD LCIA Results for the Aesthetics Impact Category

Process Group	Material	LCI Data Type	Odor (m3)	% of Total
LCD incineration	Formaldehyde	secondary	-5.17e+00	-1.03e-04
Fuel Oil #4 Prod.	Formaldehyde	secondary	-6.51e+00	-1.29e-04
Natural Gas Prod.	Acetaldehyde	secondary	-8.13e+00	-1.61e-04
LCD incineration	Ammonia	secondary	-1.35e+01	-2.68e-04
Fuel Oil #4 Prod.	Acetaldehyde	secondary	-3.36e+01	-6.67e-04
Natural Gas Prod.	Hydrogen sulfide	secondary	-5.19e+01	-1.03e-03
Natural Gas Prod.	Ammonia	secondary	-5.20e+01	-1.03e-03
LCD incineration	Propionaldehyde	secondary	-1.05e+02	-2.08e-03
LCD incineration	Hydrogen sulfide	secondary	-1.20e+03	-2.37e-02
LCD incineration	Acetaldehyde	secondary	-2.04e+03	-4.04e-02
Fuel Oil #4 Prod.	Hydrogen sulfide	secondary	-1.96e+04	-3.89e-01
Total End-of-life			-2.30e+04	-4.56e-01
Total All Life-cycle Stages			5.04e+06	1.00e+02

 Table M-36.
 LCD LCIA Results for the Aesthetics Impact Category

Process Group	Material	LCI Data Type	Aquatic Toxicity (tox-kg)	% of Total
Materials Processing Life-cycle S	Stage			
Aluminum Prod.	Aluminum (+3)	secondary	2.76e-02	1.23e+01
Aluminum Prod.	Copper (+1 & +2)	secondary	2.13e-02	9.46e+00
Invar	Copper (+1 & +2)	secondary	1.11e-02	4.96e+00
Invar	Aluminum (+3)	secondary	9.89e-03	4.40e+00
Invar	Zinc (+2)	secondary	9.00e-03	4.00e+00
Lead	Aluminum (+3)	secondary	8.03e-03	3.57e+00
Ferrite mfg.	Zinc (+2)	secondary	6.81e-03	3.03e+00
Aluminum Prod.	Zinc (+2)	secondary	6.44e-03	2.86e+00
ABS Production	Ammonia	secondary	6.17e-03	2.74e+00
Lead	Copper (+1 & +2)	secondary	6.15e-03	2.73e+00
Steel Prod., cold-rolled, semi-finished	Phosphorus (yellow or white)	secondary	4.60e-03	2.04e+00
Steel Prod., cold-rolled, semi-finished	Ammonia	secondary	2.80e-03	1.24e+00
Lead	Zinc (+2)	secondary	1.83e-03	8.16e-01
Polycarbonate Production	Copper (+1 & +2)	secondary	1.23e-03	5.45e-01
Steel Prod., cold-rolled, semi-finished	Copper (+1 & +2)	secondary	1.02e-03	4.52e-01
Ferrite mfg.	Aluminum (+3)	secondary	9.68e-04	4.31e-01
Aluminum Prod.	Barium sulfate	secondary	9.07e-04	4.03e-01
ABS Production	Aluminum (+3)	secondary	8.82e-04	3.92e-01
Invar	Ammonia	secondary	8.20e-04	3.65e-01
Ferrite mfg.	Copper (+1 & +2)	secondary	7.04e-04	3.13e-01
ABS Production	Copper (+1 & +2)	secondary	5.62e-04	2.50e-01
Styrene-butadiene Copolymer Prod.	Copper (+1 & +2)	secondary	5.49e-04	2.44e-01
Aluminum Prod.	Titanium tetrachloride	secondary	4.56e-04	2.03e-01
Polycarbonate Production	Mercury compounds	secondary	4.30e-04	1.91e-01
Aluminum Prod.	Strontium (Sr II)	secondary	3.19e-04	1.42e-01
Steel Prod., cold-rolled, semi-finished	Aluminum (+3)	secondary	2.74e-04	1.22e-01
Ferrite mfg.	Ammonia	secondary	2.64e-04	1.18e-01
Lead	Barium sulfate	secondary	2.29e-04	1.02e-01
Steel Prod., cold-rolled, semi-finished	Nitrogen dioxide	secondary	1.98e-04	8.79e-02
ABS Production	Mercury compounds	secondary	1.97e-04	8.77e-02
Steel Prod., cold-rolled, semi-finished	Zinc (+2)	secondary	1.96e-04	8.72e-02
Styrene-butadiene Copolymer Prod.	Mercury compounds	secondary	1.93e-04	8.57e-02
Steel Prod., cold-rolled, semi-finished	Fluorides (F-)	secondary	1.93e-04	8.57e-02
Aluminum Prod.	Lead cmpds	secondary	1.72e-04	7.64e-02
Polycarbonate Production	Zinc (+2)	secondary	1.71e-04	7.59e-02
Invar	Strontium (Sr II)	secondary	1.65e-04	7.36e-02
Invar	Titanium tetrachloride	secondary	1.62e-04	7.22e-02
Aluminum Prod.	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI)	secondary	1.55e-04	6.89e-02
Steel Prod., cold-rolled, semi-finished	Strontium (Sr II)	secondary	1.49e-04	6.63e-02
Lead	Titanium tetrachloride	secondary	1.33e-04	5.93e-02
Polycarbonate Production	Chlorine	secondary	1.22e-04	5.42e-02
Polycarbonate Production	Ammonia	secondary	1.19e-04	5.28e-02
Aluminum Prod.	Cadmium cmpds	secondary	1.16e-04	5.16e-02
Aluminum Prod.	Barium cmpds	secondary	9.35e-05	4.16e-02
Aluminum Prod.	Nitrate	secondary	9.13e-05	4.06e-02
Steel Prod., cold-rolled, semi-finished	Mercury compounds	secondary	8.00e-05	3.56e-02
ABS Production	Zinc (+2)	secondary	7.82e-05	3.48e-02
Styrene-butadiene Copolymer Prod.	Zinc (+2)	secondary	7.64e-05	3.40e-02
Invar	Barium sulfate	secondary	6.91e-05	3.08e-02

Table M-37. CRT LCIA Results for the Aquatic Toxicity Impact Category

Process Group	Material	LCI Data Type	Aquatic Toxicity (tox-kg)	% of Total
Steel Prod., cold-rolled, semi-finished	Barium sulfate	secondary	6.87e-05	3.05e-02
Ferrite mfg.	Barium sulfate	secondary	6.75e-05	3.00e-02
Steel Prod., cold-rolled, semi-finished	Xylene (mixed isomers)	secondary	6.51e-05	2.89e-02
Lead	Strontium (Sr II)	secondary	6.40e-05	2.84e-02
ABS Production	Nitrate	secondary	6.01e-05	2.67e-02
Invar	Lead cmpds	secondary	5.87e-05	2.61e-02
ABS Production	Chlorine	secondary	5.59e-05	2.49e-02
Polycarbonate Production	Phenol	secondary	5.55e-05	2.47e-02
Styrene-butadiene Copolymer Prod.	Chlorine	secondary	5.46e-05	2.43e-02
Lead	Lead cmpds	secondary	5.41e-05	2.41e-02
Invar	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI)	secondary	5.40e-05	2.40e-02
Polystyrene Prod., high-impact	Ammonia	secondary	5.18e-05	2.30e-02
Ferrite mfg.	Strontium (Sr II)	secondary	4.84e-05	2.15e-02
Lead	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI)	secondary	4.68e-05	2.08e-02
Invar	Barium cmpds	secondary	3.86e-05	1.72e-02
Aluminum Prod.	Nickel cmpds	secondary	3.82e-05	1.70e-02
Aluminum Prod.	Chromium (III)	secondary	3.20e-05	1.42e-02
Invar	Nickel cmpds	secondary	3.03e-05	1.35e-02
Steel Prod., cold-rolled, semi-finished	Silver compounds	secondary	2.83e-05	1.26e-02
Aluminum Prod.	Aromatic hydrocarbons	secondary	2.83e-05	1.26e-02
Invar	Xylene (mixed isomers)	secondary	2.44e-05	1.08e-02
Lead	Barium cmpds	secondary	2.43e-05	1.08e-02
Aluminum Prod.	Manganese cmpds	secondary	2.28e-05	1.01e-02
Ferrite mfg.	Phosphorus (yellow or white)	secondary	2.21e-05	9.85e-03
Invar	Manganese cmpds	secondary	2.10e-05	9.36e-03
Ferrite mfg.	Xylene (mixed isomers)	secondary	1.93e-05	8.56e-03
Invar	Phosphorus (yellow or white)	secondary	1.86e-05	8.29e-03
Aluminum Prod.	Fluoride	secondary	1.82e-05	8.09e-03
Lead	Nitrate	secondary	1.80e-05	8.01e-03
Steel Prod., cold-rolled, semi-finished	Lead cmpds	secondary	1.74e-05	7.74e-03
Aluminum Prod.	Vanadium (V3+, V5+)	secondary	1.63e-05	7.24e-03
Aluminum Prod.	Selenium	secondary	1.59e-05	7.05e-03
Steel Prod., cold-rolled, semi-finished	Barium cmpds	secondary	1.55e-05	6.90e-03
Ferrite mfg.	Titanium tetrachloride	secondary	1.50e-05	6.66e-03
Invar	Aromatic hydrocarbons	secondary	1.47e-05	6.52e-03
Styrene-butadiene Copolymer Prod.	Aluminum (+3)	secondary	1.44e-05	6.39e-03
Ferrite mfg.	Manganese cmpds	secondary	1.32e-05	5.87e-03
Steel Prod., cold-rolled, semi-finished	Aromatic hydrocarbons	secondary	1.30e-05	5.78e-03
Aluminum Prod.	Xylene (mixed isomers)	secondary	1.26e-05	5.61e-03
Invar	Cadmium cmpds	secondary	1.26e-05	5.60e-03
Invar	Chromium (III)	secondary	1.14e-05	5.09e-03
Lead	Nickel cmpds	secondary	1.12e-05	4.99e-03
Steel Prod., cold-rolled, semi-finished	Tin (Sn++, Sn4+)	secondary	1.04e-05	4.63e-03
Ferrite mfg.	Lead cmpds	secondary	1.02e-05	4.55e-03
Lead	Chromium (III)	secondary	9.38e-06	4.17e-03
Lead	Fluoride	secondary	8.91e-06	3.96e-03
Invar	Fluoride	secondary	8.80e-06	3.91e-03
Aluminum Prod.	Arsenic cmpds	secondary	8.57e-06	3.81e-03
Invar	Silver compounds	secondary	8.56e-06	3.81e-03
Ferrite mfg.	Silver compounds	secondary	8.35e-06	3.72e-03

Process Group	Material	LCI Data Type	Aquatic Toxicity (tox-kg)	% of Total
Ferrite mfg.	Nitrate	secondary	8.08e-06	3.59e-03
Polycarbonate Production	Aluminum (+3)	secondary	8.02e-06	3.56e-03
Ferrite mfg.	Barium cmpds	secondary	7.00e-06	3.11e-03
Aluminum Prod.	Benzene	secondary	6.79e-06	3.02e-03
Lead	Manganese cmpds	secondary	6.47e-06	2.88e-03
Invar	Toluene	secondary	6.23e-06	2.77e-03
Invar	Vanadium (V3+, V5+)	secondary	6.00e-06	2.67e-03
Styrene-butadiene Copolymer Prod.	Nitrate	secondary	5.79e-06	2.58e-03
Invar	Selenium	secondary	5.69e-06	2.53e-03
Lead	Vanadium (V3+, V5+)	secondary	4.75e-06	2.11e-03
Lead	Selenium	secondary	4.63e-06	2.06e-03
Aluminum Prod.	Toluene	secondary	4.24e-06	1.89e-03
Ferrite mfg.	Aromatic hydrocarbons	secondary	4.10e-06	1.82e-03
Aluminum Prod.	Phenol	secondary	3.79e-06	1.69e-03
Invar	Benzene	secondary	3.63e-06	1.61e-03
Steel Prod., cold-rolled, semi-finished	Phenol	secondary	3.55e-06	1.58e-03
Steel Prod., cold-rolled, semi-finished	Benzene	secondary	3.54e-06	1.58e-03
ABS Production	Phenol	secondary	3.49e-06	1.55e-03
Invar	Arsenic cmpds	secondary	3.05e-06	1.35e-03
Invar	Tin (Sn++, Sn4+)	secondary	3.03e-06	1.35e-03
Steel Prod., cold-rolled, semi-finished	Chlorine	secondary	3.01e-06	1.34e-03
Styrene-butadiene Copolymer Prod.	Phenol	secondary	2.93e-06	1.30e-03
Invar	Cobalt (Co I, Co II, Co III)	secondary	2.88e-06	1.28e-03
Steel Prod., cold-rolled, semi-finished	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI)	secondary	2.84e-06	1.26e-03
Steel Prod., cold-rolled, semi-finished	Cadmium cmpds	secondary	2.68e-06	1.19e-03
Lead	Arsenic cmpds	secondary	2.49e-06	1.11e-03
Lead	Xylene (mixed isomers)	secondary	2.44e-06	1.08e-03
Steel Prod., cold-rolled, semi-finished	Cyanide (-1)	secondary	2.37e-06	1.05e-03
Steel Prod., cold-rolled, semi-finished	Boric acid	secondary	2.27e-06	1.01e-03
Steel Prod., cold-rolled, semi-finished	Toluene	secondary	2.22e-06	9.87e-04
Steel Prod., cold-rolled, semi-finished	Manganese cmpds	secondary	2.10e-06	9.32e-04
Invar	Phenol	secondary	1.99e-06	8.85e-04
Ferrite mfg.	Fluoride	secondary	1.85e-06	8.21e-04
Lead	Cobalt (Co I, Co II, Co III)	secondary	1.84e-06	8.16e-04
Invar	Ethylbenzene	secondary	1.77e-06	7.87e-04
Steel Prod., cold-rolled, semi-finished	Ethylbenzene	secondary	1.76e-06	7.81e-04
Ferrite mfg.	Nickel cmpds	secondary	1.60e-06	7.10e-04
Lead	Mercury compounds	secondary	1.39e-06	6.20e-04
Lead	Benzene	secondary	1.32e-06	5.86e-04
Aluminum Prod.	Nitrites	secondary	1.25e-06	5.58e-04
Steel Prod., cold-rolled, semi-finished	Chromium (VI)	secondary	1.23e-06	5.48e-04
Steel Prod., cold-rolled, semi-finished	Nickel cmpds	secondary	1.19e-06	5.31e-04
Ferrite mfg.	Benzene	secondary	1.08e-06	4.82e-04
Ferrite mfg.	Chromium (III)	secondary	1.08e-06	4.81e-04
Polystyrene Prod., high-impact	Phenol	secondary	1.07e-06	4.75e-04
ABS Production	Nickel cmpds	secondary	1.01e-06	4.49e-04
Polycarbonate Production	Nickel (+2)	secondary	9.23e-07	4.11e-04
Polycarbonate Production	Nitrate	secondary	9.23e-07	4.11e-04
Lead	Toluene	secondary	8.22e-07	3.66e-04
Lead	Phenol	secondary	8.16e-07	3.63e-04
		5		

Table M-37. CRT LCIA Results for the Aquatic Toxicity Impact Category

Process Group	Material	LCI Data Type	Aquatic Toxicity (tox-kg)	% of Total
Steel Prod., cold-rolled, semi-finished	Chromium (III)	secondary	8.02e-07	3.57e-04
Ferrite mfg.	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI)	secondary	7.99e-07	3.55e-04
Ferrite mfg.	Vanadium (V3+, V5+)	secondary	6.83e-07	3.04e-04
Ferrite mfg.	Toluene	secondary	6.82e-07	3.04e-04
Invar	Boric acid	secondary	6.44e-07	2.86e-04
Ferrite mfg.	Boric acid	secondary	6.29e-07	2.80e-04
Polystyrene Prod., high-impact	Nitrate	secondary	6.05e-07	2.69e-04
Steel Prod., cold-rolled, semi-finished	Vanadium (V3+, V5+)	secondary	5.94e-07	2.64e-04
Ferrite mfg.	Cadmium cmpds	secondary	5.89e-07	2.62e-04
Invar	Cyanide (-1)	secondary	5.64e-07	2.51e-04
Invar	Nitrites	secondary	5.52e-07	2.46e-04
Ferrite mfg.	Selenium	secondary	5.52e-07	2.45e-04
Ferrite mfg.	Phenol	secondary	5.50e-07	2.45e-04
Steel Prod., cold-rolled, semi-finished	Titanium tetrachloride	secondary	5.28e-07	2.35e-04
Ferrite mfg.	Ethylbenzene	secondary	5.20e-07	2.31e-04
Invar	Nitrates/nitrites	secondary	4.69e-07	2.09e-04
Styrene-butadiene Copolymer Prod.	Nickel (+2)	secondary	4.14e-07	1.84e-04
Ferrite mfg.	Cyanide (-1)	secondary	3.97e-07	1.77e-04
Aluminum Prod.	Cyanide (-1)	secondary	3.79e-07	1.69e-04
Invar	Boron (B III)	secondary	3.27e-07	1.46e-04
Ferrite mfg.	Arsenic cmpds	secondary	2.96e-07	1.31e-04
Steel Prod., cold-rolled, semi-finished	Arsenic cmpds	secondary	2.77e-07	1.23e-04
Lead	Nitrites	secondary	2.65e-07	1.18e-04
Invar	Chlorine	secondary	2.51e-07	1.12e-04
Ferrite mfg.	Chlorine	secondary	2.45e-07	1.09e-04
Steel Prod., cold-rolled, semi-finished	Rubidium ion (Rb+)	secondary	2.42e-07	1.08e-04
Lead	Boron (B III)	secondary	1.97e-07	8.74e-05
Steel Prod., cold-rolled, semi-finished	Nitrates/nitrites	secondary	1.94e-07	8.63e-05
Ferrite mfg.	Cobalt (Co I, Co II, Co III)	secondary	1.76e-07	7.82e-05
Steel Prod., cold-rolled, semi-finished	Selenium	secondary	1.59e-07	7.09e-05
Invar	Mercury compounds	secondary	9.86e-08	4.39e-05
Ferrite mfg.	Mercury compounds	secondary	9.63e-08	4.28e-05
Invar	Rubidium ion (Rb+)	secondary	7.31e-08	3.25e-05
Ferrite mfg.	Rubidium ion (Rb+)	secondary	7.14e-08	3.17e-05
Steel Prod., cold-rolled, semi-finished	Boron (B III)	secondary	7.04e-08	3.13e-05
Ferrite mfg.	Nitrites	secondary	3.83e-08	1.71e-05
Invar	Chromium (VI)	secondary	3.25e-08	1.45e-05
Steel Prod., cold-rolled, semi-finished	Phosphorus pentoxide	secondary	3.11e-08	1.38e-05
Ferrite mfg.	Chromium (VI)	secondary	2.76e-08	1.23e-05
Ferrite mfg.	Boron (B III)	secondary	2.32e-08	1.03e-05
Steel Prod., cold-rolled, semi-finished	Hydrazine	secondary	2.30e-08	1.02e-05
Steel Prod., cold-rolled, semi-finished	Morpholine	secondary	1.88e-08	8.35e-06
Aluminum Prod.	Chromium (VI)	secondary	9.21e-09	4.09e-06
Invar	Phosphorus pentoxide	secondary	9.09e-09	4.04e-06
Ferrite mfg.	Phosphorus pentoxide	secondary	8.87e-09	3.94e-06
Invar	Hydrazine	secondary	6.52e-09	2.90e-06
Ferrite mfg.	Hydrazine	secondary	6.37e-09	2.83e-06
Invar	Morpholine	secondary	5.32e-09	2.36e-06
Ferrite mfg.	Morpholine	secondary	5.19e-09	2.31e-06
Lead	Chromium (VI)	secondary	4.46e-09	1.99e-06

 Table M-37. CRT LCIA Results for the Aquatic Toxicity Impact Category

Process Group	Material	LCI Data Type	Aquatic Toxicity (tox-kg)	% of Total
Invar	Hypochlorous acid	secondary	3.54e-09	1.58e-06
Steel Prod., cold-rolled, semi-finished	Hypochlorous acid	secondary	3.52e-09	1.56e-06
Ferrite mfg.	Hypochlorous acid	secondary	3.46e-09	1.54e-06
Steel Prod., cold-rolled, semi-finished	Cobalt (Co I, Co II, Co III)	secondary	3.14e-09	1.40e-06
Invar	Dichloromethane	secondary	2.00e-09	8.88e-07
Steel Prod., cold-rolled, semi-finished	Dichloromethane	secondary	1.98e-09	8.82e-07
Ferrite mfg.	Dichloromethane	secondary	1.95e-09	8.67e-07
Aluminum Prod.	Triethylene glycol	secondary	1.73e-09	7.71e-07
Invar	Acetic acid	secondary	1.51e-09	6.73e-07
Ferrite mfg.	Acetic acid	secondary	1.48e-09	6.57e-07
Invar	Triethylene glycol	secondary	1.38e-09	6.14e-07
Steel Prod., cold-rolled, semi-finished	Triethylene glycol	secondary	1.37e-09	6.09e-07
Ferrite mfg.	Triethylene glycol	secondary	1.35e-09	5.99e-07
Lead	Triethylene glycol	secondary	1.30e-09	5.76e-07
Steel Prod., cold-rolled, semi-finished	Nitrites	secondary	4.82e-10	2.14e-07
Invar	Halogenated matter (organic)	secondary	2.74e-10	1.22e-07
Ferrite mfg.	Halogenated matter (organic)	secondary	2.68e-10	1.19e-07
Steel Prod., cold-rolled, semi-finished	Edetic acid (EDTA)	secondary	1.27e-10	5.66e-08
Invar	Edetic acid (EDTA)	secondary	3.60e-11	1.60e-08
Ferrite mfg.	Edetic acid (EDTA)	secondary	3.52e-11	1.56e-08
Ferrite mfg.	Tin (Sn++, Sn4+)	secondary	7.28e-12	3.24e-09
Invar	Chloroform	secondary	3.23e-12	1.44e-09
Steel Prod., cold-rolled, semi-finished	Chloroform	secondary	3.21e-12	1.43e-09
Ferrite mfg.	Chloroform	secondary	3.15e-12	1.40e-09
Steel Prod., cold-rolled, semi-finished	Lithium salts	secondary	2.38e-12	1.06e-09
Invar	Trichloroethylene	secondary	1.61e-12	7.16e-10
Steel Prod., cold-rolled, semi-finished	Trichloroethylene	secondary	1.60e-12	7.10e-10
Ferrite mfg.	Trichloroethylene	secondary	1.57e-12	6.99e-10
Invar	Lithium salts	secondary	6.75e-13	3.00e-10
Ferrite mfg.	Lithium salts	secondary	6.59e-13	2.93e-10
Invar	Formaldehyde	secondary	1.22e-13	5.42e-11
Ferrite mfg.	Formaldehyde	secondary	1.19e-13	5.29e-11
Steel Prod., cold-rolled, semi-finished	Formaldehyde	secondary	1.18e-13	5.23e-11
Steel Prod., cold-rolled, semi-finished	Tetrachloroethylene	secondary	8.60e-14	3.82e-11
Invar	Tetrachloroethylene	secondary	8.47e-14	3.77e-11
Ferrite mfg.	Tetrachloroethylene	secondary	8.27e-14	3.68e-11
Invar	1,1,1-Trichloroethane	secondary	6.01e-14	2.67e-11
Steel Prod., cold-rolled, semi-finished	1,1,1-Trichloroethane	secondary	5.94e-14	2.64e-11
Ferrite mfg.	1,1,1-Trichloroethane	secondary	5.86e-14	2.60e-11
Invar	Hexachloroethane	secondary	3.60e-16	1.60e-13
Steel Prod., cold-rolled, semi-finished	Hexachloroethane	secondary	3.58e-16	1.59e-13
Ferrite mfg.	Hexachloroethane	secondary	3.52e-16	1.56e-13
Total Materials Processing			1.36e-01	6.04e+01
Manufacturing Life-cycle Stage				
CRT tube mfg.	Phosphorus (yellow or white)	primary	5.94e-02	2.64e+01
CRT tube mfg.	Fluoride	primary	6.91e-03	3.07e+00
Glass/frit	Fluorides (F-)	primary	5.87e-03	2.61e+00
CRT tube mfg.	Zinc (elemental)	primary	5.12e-03	2.28e+00
CRT tube mfg.	Copper	primary	4.78e-03	2.13e+00

Table M-37. CRT LCIA Results for the Aquatic Toxicity Impact Category

Process Group	Material	LCI Data Type	Aquatic Toxicity (tox-kg)	% of Total
LPG Production	Phenol	secondary	4.25e-03	1.89e+00
LPG Production	Aluminum (+3)	secondary	2.46e-03	1.09e+00
CRT tube mfg.	Nickel	primary	1.59e-04	7.05e-02
LPG Production	Nitrate	secondary	1.21e-04	5.39e-02
Glass/frit	Lead	primary	8.68e-05	3.86e-02
LPG Production	Fluorides (F-)	secondary	6.81e-05	3.03e-02
LPG Production	Copper (+1 & +2)	secondary	1.55e-05	6.89e-03
Fuel Oil #6 Prod.	Phenol	secondary	1.44e-05	6.39e-03
Fuel Oil #2 Prod.	Phenol	secondary	1.19e-05	5.30e-03
Fuel Oil #6 Prod.	Aluminum (+3)	secondary	8.32e-06	3.70e-03
CRT tube mfg.	Manganese	primary	7.21e-06	3.21e-03
Fuel Oil #2 Prod.	Aluminum (+3)	secondary	6.90e-06	3.07e-03
CRT tube mfg.	Lead	primary	6.03e-06	2.68e-03
LPG Production	Zinc (+2)	secondary	4.34e-06	1.93e-03
CRT tube mfg.	Molybdenum	primary	3.77e-06	1.68e-03
CRT tube mfg.	Cyanide (-1)	primary	1.21e-06	5.39e-04
Fuel Oil #4 Prod.	Phenol	secondary	9.69e-07	4.31e-04
Fuel Oil #4 Prod.	Aluminum (+3)	secondary	5.61e-07	2.49e-04
Fuel Oil #6 Prod.	Fluorides (F-)	secondary	5.50e-07	2.44e-04
Fuel Oil #6 Prod.	Nitrate	secondary	4.85e-07	2.16e-04
Fuel Oil #2 Prod.	Nitrate	secondary	3.45e-07	1.54e-04
Fuel Oil #2 Prod.	Fluorides (F-)	secondary	2.14e-07	9.50e-05
Natural Gas Prod.	Zinc (+2)	secondary	1.94e-07	8.61e-05
LPG Production	Aromatic hydrocarbons	secondary	1.75e-07	7.78e-05
Glass/frit	Nickel	primary	1.64e-07	7.30e-05
LPG Production	Barium cmpds	secondary	1.42e-07	6.32e-05
Natural Gas Prod.	Fluorides (F-)	secondary	1.19e-07	5.30e-05
Glass/frit	Nitrates/nitrites	primary	1.14e-07	5.09e-05
LPG Production	Chromium (VI)	secondary	1.02e-07	4.52e-05
Glass/frit	Chromium	primary	9.86e-08	4.39e-05
Natural Gas Prod.	Phenol	secondary	8.65e-08	3.85e-05
LPG Production	Chromium (III)	secondary	7.28e-08	3.24e-05
LPG Production	Cadmium cmpds	secondary	6.89e-08	3.06e-05
Fuel Oil #6 Prod.	Copper $(+1 \& +2)$	secondary	5.24e-08	2.33e-05
Natural Gas Prod.	Aluminum (+3)	secondary	4.97e-08	2.21e-05
Fuel Oil #2 Prod.	Copper $(+1 \& +2)$	secondary	4.34e-08	1.93e-05
Natural Gas Prod.	Nitrate	secondary	3.01e-08	1.34e-05
Fuel Oil #4 Prod.	Nitrate	secondary	2.93e-08	1.31e-05
LPG Production	Lead cmpds	secondary	2.30e-08	1.02e-05
Fuel Oil #4 Prod	Fluorides (F-)	secondary	2.28e-08	1.01e-05
LPG Production	Toluene	secondary	2.14e-08	9.51e-06
Fuel Oil #6 Prod	$Z_{inc}(+2)$	secondary	1 97e-08	8 75e-06
Fuel Oil #2 Prod	$Z_{\text{inc}}(+2)$	secondary	1.25e-08	5 57e-06
Fuel Oil #4 Prod	Copper $(+1 \& +2)$	secondary	3 53e-09	1.57e-06
Natural Gas Prod	Chromium (VI)	secondary	3.06e-09	1.37°C 00
LPG Production	Nickel cmpds	secondary	2 78e-09	1.23e-06
Natural Gas Prod	Chromium (III)	secondary	2.700-09 2.20e-00	9.77e-07
LPG Production	Mercury compounds	secondary	1 25e-09	5 56e-07
Fuel Oil #4 Prod	Zinc (+2)	secondary	1.250-09	4 90e-07
I PG Production	Cyanide (-1)	secondary	8 17e 10	3 630 07
	Cyaniac (-1)	secondary	0.1/0-10	5.056-07

Process Group	Material	LCI Data Type	Aquatic Toxicity (tox-kg)	% of Total
Fuel Oil #6 Prod.	Aromatic hydrocarbons	secondary	5.92e-10	2.63e-07
Fuel Oil #2 Prod.	Aromatic hydrocarbons	secondary	4.91e-10	2.18e-07
Fuel Oil #6 Prod.	Barium cmpds	secondary	4.81e-10	2.14e-07
Fuel Oil #2 Prod.	Barium cmpds	secondary	3.99e-10	1.77e-07
Natural Gas Prod.	Copper (+1 & +2)	secondary	3.13e-10	1.39e-07
Fuel Oil #6 Prod.	Chromium (VI)	secondary	2.51e-10	1.12e-07
LPG Production	Halogenated matter (organic)	secondary	2.33e-10	1.04e-07
Fuel Oil #6 Prod.	Cadmium cmpds	secondary	2.33e-10	1.04e-07
Fuel Oil #2 Prod.	Cadmium cmpds	secondary	1.93e-10	8.59e-08
Fuel Oil #6 Prod.	Chromium (III)	secondary	1.80e-10	8.01e-08
Fuel Oil #2 Prod.	Chromium (VI)	secondary	1.48e-10	6.58e-08
Fuel Oil #2 Prod.	Chromium (III)	secondary	1.06e-10	4.72e-08
Fuel Oil #6 Prod.	Lead cmpds	secondary	7.78e-11	3.46e-08
Fuel Oil #6 Prod.	Toluene	secondary	7.23e-11	3.22e-08
Fuel Oil #2 Prod.	Lead cmpds	secondary	6.45e-11	2.87e-08
Fuel Oil #2 Prod.	Toluene	secondary	6.00e-11	2.67e-08
Fuel Oil #4 Prod.	Aromatic hydrocarbons	secondary	3.99e-11	1.78e-08
Fuel Oil #4 Prod.	Barium cmpds	secondary	3.24e-11	1.44e-08
Fuel Oil #4 Prod.	Cadmium cmpds	secondary	1.57e-11	6.98e-09
Fuel Oil #4 Prod.	Chromium (VI)	secondary	1.34e-11	5.95e-09
Fuel Oil #4 Prod.	Chromium (III)	secondary	9.60e-12	4.27e-09
Fuel Oil #6 Prod.	Nickel cmpds	secondary	9.42e-12	4.19e-09
Fuel Oil #2 Prod.	Nickel cmpds	secondary	7.81e-12	3.47e-09
Fuel Oil #4 Prod.	Lead cmpds	secondary	5.24e-12	2.33e-09
Fuel Oil #4 Prod.	Toluene	secondary	4.88e-12	2.17e-09
Fuel Oil #6 Prod.	Mercury compounds	secondary	4.23e-12	1.88e-09
Natural Gas Prod.	Aromatic hydrocarbons	secondary	3.54e-12	1.57e-09
Fuel Oil #2 Prod.	Mercury compounds	secondary	3.51e-12	1.56e-09
Natural Gas Prod.	Barium cmpds	secondary	2.87e-12	1.28e-09
Fuel Oil #6 Prod.	Cyanide (-1)	secondary	2.76e-12	1.23e-09
Fuel Oil #2 Prod.	Cyanide (-1)	secondary	2.29e-12	1.02e-09
Natural Gas Prod.	Cadmium cmpds	secondary	1.39e-12	6.19e-10
Fuel Oil #6 Prod.	Halogenated matter (organic)	secondary	7.89e-13	3.51e-10
Fuel Oil #2 Prod.	Halogenated matter (organic)	secondary	6.55e-13	2.91e-10
Fuel Oil #4 Prod.	Nickel cmpds	secondary	6.35e-13	2.82e-10
Natural Gas Prod.	Lead cmpds	secondary	4.65e-13	2.07e-10
Natural Gas Prod.	Toluene	secondary	4.32e-13	1.92e-10
Fuel Oil #4 Prod.	Mercury compounds	secondary	2.85e-13	1.27e-10
Fuel Oil #4 Prod.	Cyanide (-1)	secondary	1.86e-13	8.28e-11
Natural Gas Prod.	Nickel cmpds	secondary	5.63e-14	2.50e-11
Fuel Oil #4 Prod.	Halogenated matter (organic)	secondary	5.32e-14	2.37e-11
Natural Gas Prod.	Mercury compounds	secondary	2.53e-14	1.12e-11
Natural Gas Prod.	Cyanide (-1)	secondary	1.65e-14	7.34e-12
Natural Gas Prod.	Halogenated matter (organic)	secondary	4.72e-15	2.10e-12
Total Manufacturing			8.93e-02	3.97e+01
Use, Maintenance and Repair Li	fe-cycle Stage			
Total Use, Maintenance and Rep	air		0.00e+00	0.00e+00
End-of-life Life-cycle Stage				
CRT landfilling	Ammonia	primary	7.71e-05	3.43e-02

Table M-37. CRT LCIA Results for the Aquatic Toxicity Impact Category

Process Group	Material	LCI Data Type	Aquatic Toxicity (tox-kg)	% of Total
CRT Incineration	Silver compounds	secondary	5.15e-06	2.29e-03
CRT landfilling	Silver compounds	primary	4.80e-06	2.14e-03
CRT Incineration	Barium cmpds	secondary	1.14e-07	5.07e-05
CRT landfilling	Cadmium cmpds	primary	9.64e-08	4.29e-05
CRT Incineration	Cadmium cmpds	secondary	9.33e-08	4.15e-05
CRT landfilling	Barium cmpds	primary	9.07e-08	4.03e-05
LPG Production	Phenol	secondary	3.67e-08	1.63e-05
CRT Incineration	Chromium (VI)	secondary	2.19e-08	9.72e-06
LPG Production	Aluminum (+3)	secondary	2.12e-08	9.44e-06
CRT landfilling	Mercury compounds	primary	2.04e-08	9.07e-06
CRT Incineration	Mercury compounds	secondary	2.00e-08	8.89e-06
CRT landfilling	Arsenic cmpds	primary	1.84e-08	8.20e-06
CRT Incineration	Arsenic cmpds	secondary	1.77e-08	7.89e-06
CRT landfilling	Chromium (VI)	primary	1.66e-08	7.39e-06
CRT Incineration	Lead cmpds	secondary	1.60e-08	7.10e-06
CRT Incineration	Chromium (III)	secondary	1.57e-08	6.97e-06
CRT landfilling	Lead cmpds	primary	1.56e-08	6.92e-06
CRT landfilling	Chromium (III)	primary	1.19e-08	5.30e-06
CRT landfilling	Xylene (mixed isomers)	primary	4.17e-09	1.86e-06
CRT Incineration	o-xylene	secondary	2.68e-09	1.19e-06
CRT Incineration	Toluene	secondary	2.53e-09	1.12e-06
CRT landfilling	Toluene	primary	2.36e-09	1.05e-06
LPG Production	Nitrate	secondary	1.05e-09	4.65e-07
CRT Incineration	Ethylbenzene	secondary	9.51e-10	4.23e-07
CRT landfilling	Ethylbenzene	primary	8.86e-10	3.94e-07
LPG Production	Fluorides (F-)	secondary	5.88e-10	2.61e-07
CRT Incineration	Tetrachloroethylene	secondary	1.45e-10	6.45e-08
CRT landfilling	Tetrachloroethylene	primary	1.35e-10	6.02e-08
LPG Production	Copper $(+1 \& +2)$	secondary	1.34e-10	5.94e-08
CRT Incineration	Benzene	secondary	9.63e-11	4.28e-08
CRT landfilling	Benzene	primary	8.98e-11	3.99e-08
CRT Incineration	Selenium	secondary	8.22e-11	3.66e-08
CRT landfilling	Selenium	primary	8.08e-11	3.59e-08
CRT Incineration	Carbon tetrachloride	secondary	5.89e-11	2.62e-08
CRT landfilling	Carbon tetrachloride	primary	5.49e-11	2.44e-08
CRT Incineration	Trichloroethylene	secondary	4.45e-11	1.98e-08
CRT landfilling	Trichloroethylene	primary	4.15e-11	1.84e-08
LPG Production	Zinc (+2)	secondary	3.74e-11	1.66e-08
CRT Incineration	Chloroform	secondary	2.38e-11	1.06e-08
CRT Incineration	Vinyl chloride	secondary	2.38e-11	1.06e-08
CRT landfilling	Chloroform	primary	2.22e-11	9.89e-09
CRT landfilling	Vinyl chloride	primary	2.22e-11	9.86e-09
CRT Incineration	1 2-Dichloroethane	secondary	1 25e-11	5 57e-09
CRT landfilling	1.2-Dichloroethane	primary	1 17e-11	5 19e-09
CRT Incineration	Dichloromethane	secondary	8 25e-12	3.67e-09
CRT landfilling	Dichloromethane	primary	7 69e-12	3 42e-09
LPG Production	Aromatic hydrocarbons	secondary	1.51e-12	6 72e-10
LPG Production	Barium cmpds	secondary	1.23e-12	5.46e-10
LPG Production	Chromium (VI)	secondary	8 76e-13	3 90e-10
LPG Production	Chromium (III)	secondary	6 29e-13	2.80e-10
			0.270 15	2.000 10

LPG ProductionCadmium empdssecondary5.94e-132.64e-10LPG ProductionLead empdssecondary1.98e-138.82e-11LPG ProductionToluenesecondary1.85e-138.21e-11LPG ProductionNickel empdssecondary2.40e-141.07e-11LPG ProductionMercury compoundssecondary1.08e-144.80e-12LPG ProductionCyanide (-1)secondary7.05e-158.14e-12LPG ProductionHalogenated matter (organic)secondary2.01e-158.96e-13Natural Gas Prod.Halogenated matter (organic)secondary-2.37e-15-1.05e-12Natural Gas Prod.Cyanide (-1)secondary-2.82e-14-5.64e-12Natural Gas Prod.Mercury compoundssecondary-2.82e-14-1.25e-11CRT IncinerationHalogenated matter (organic)secondary-2.37e-13-9.64e-11Natural Gas Prod.Cyanide (-1)secondary-2.37e-14-5.64e-12Natural Gas Prod.Nickel empdssecondary-1.29e-13-5.71e-11RT IncinerationCyanide (-1)secondary-2.33e-13-1.04e-10RT IncinerationCyanide (mpdssecondary-2.33e-13-1.04e-10RT IncinerationNickel empdssecondary-2.33e-13-1.04e-10RT IncinerationNickel empdssecondary-5.72e-13-2.54e-10Ratural Gas Prod.Lead empdssecondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium empds <th>Process Group</th> <th>Material</th> <th>LCI Data Type</th> <th>Aquatic Toxicity (tox-kg)</th> <th>% of Total</th>	Process Group	Material	LCI Data Type	Aquatic Toxicity (tox-kg)	% of Total
LPG ProductionLead empdssecondary1.98e-138.82e-14LPG ProductionToluenesecondary1.85e-138.21e-11LPG ProductionNickel empdssecondary2.40e-141.07e-11LPG ProductionMercury compoundssecondary7.05e-153.14e-12LPG ProductionCyanide (-1)secondary7.05e-153.14e-12LPG ProductionHalogenated matter (organic)secondary2.01e-158.96e-13Natural Gas Prod.Halogenated matter (organic)secondary-2.37e-15-1.05e-12Natural Gas Prod.Cyanide (-1)secondary-8.28e-15-3.68e-12Natural Gas Prod.Mercury compoundssecondary-1.27e-14-5.64e-12Natural Gas Prod.Nickel empdssecondary-2.37e-15-1.15e-112CRT IncinerationHalogenated matter (organic)secondary-2.32e-14-1.25e-11CRT IncinerationCyanide (-1)secondary-2.32e-14-1.63e-11Natural Gas Prod.Lead empdssecondary-2.32e-13-5.71e-11Natural Gas Prod.Lead empdssecondary-2.32e-13-5.71e-11Natural Gas Prod.Lead empdssecondary-2.32e-13-5.71e-10Natural Gas Prod.Lead empdssecondary-2.32e-13-3.10e-10CRT IncinerationNickel empdssecondary-2.32e-13-2.54e-10Natural Gas Prod.Lead empdssecondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium empds </td <td>LPG Production</td> <td>Cadmium cmpds</td> <td>secondary</td> <td>5.94e-13</td> <td>2.64e-10</td>	LPG Production	Cadmium cmpds	secondary	5.94e-13	2.64e-10
LPG ProductionToluenesecondary1.85e-138.21e-11LPG ProductionNickel empdssecondary2.40e-141.07e-11LPG ProductionMercury compoundssecondary1.08e-144.80e-12LPG ProductionCyanide (-1)secondary7.05e-153.14e-12LPG ProductionHalogenated matter (organic)secondary2.01e-158.96e-13Natural Gas Prod.Halogenated matter (organic)secondary-2.37e-15-1.05e-12Natural Gas Prod.Cyanide (-1)secondary-8.28e-15-3.68e-12Natural Gas Prod.Mercury compoundssecondary-2.82e-14-1.25e-11Natural Gas Prod.Nickel empdssecondary-2.82e-14-1.63e-11CRT IncinerationHalogenated matter (organic)secondary-3.67e-14-1.63e-11CRT IncinerationCyanide (-1)secondary-2.37e-13-5.71e-11Natural Gas Prod.Lead empdssecondary-2.32e-13-5.71e-11Natural Gas Prod.Lead empdssecondary-2.32e-13-5.71e-11Natural Gas Prod.Lead empdssecondary-2.33e-13-1.04e-10CRT IncinerationNickel empdssecondary-2.32e-13-5.71e-11Natural Gas Prod.Lead empdssecondary-2.32e-13-3.10e-10CRT IncinerationNickel empdssecondary-2.32e-13-3.10e-10Fuel Oil #4 Prod.Halogenated matter (organic)secondary-3.72e-13-3.25e-10Natural Gas Prod.<	LPG Production	Lead cmpds	secondary	1.98e-13	8.82e-11
LPG ProductionNickel empdssecondary2.40e-141.07e-11LPG ProductionMercury compoundssecondary1.08e-144.80e-12LPG ProductionCyanide (-1)secondary7.05e-153.14e-12LPG ProductionHalogenated matter (organic)secondary2.01e-158.96e-13Natural Gas Prod.Halogenated matter (organic)secondary-2.37e-15-1.05e-12Natural Gas Prod.Cyanide (-1)secondary-8.28e-15-3.68e-12Natural Gas Prod.Mercury compoundssecondary-2.82e-14-1.25e-11CRT IncinerationHalogenated matter (organic)secondary-2.82e-14-1.63e-11CRT IncinerationCyanide (-1)secondary-3.67e-14-1.63e-11Natural Gas Prod.Cyanide (-1)secondary-2.37e-15-7.1ce-11RT IncinerationCyanide (-1)secondary-2.32e-14-1.25e-11CRT IncinerationCyanide (-1)secondary-2.37e-13-5.71e-11Natural Gas Prod.Lead empdssecondary-2.33e-13-1.04e-10CRT IncinerationNickel empdssecondary-2.33e-13-1.04e-10CRT IncinerationNickel empdssecondary-5.72e-13-2.54e-10Natural Gas Prod.Lead empdssecondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium empdssecondary-5.72e-13-2.54e-10Natural Gas Prod.Gamium empdssecondary-1.44e-12-6.44e-10Natural Gas Prod.	LPG Production	Toluene	secondary	1.85e-13	8.21e-11
LPG ProductionMercury compoundssecondary1.08e-144.80e-12LPG ProductionCyanide (-1)secondary7.05e-153.14e-12LPG ProductionHalogenated matter (organic)secondary2.01e-158.96e-13Natural Gas Prod.Halogenated matter (organic)secondary-2.37e-15-1.05e-12Natural Gas Prod.Cyanide (-1)secondary-8.28e-15-3.68e-12Natural Gas Prod.Mercury compoundssecondary-1.27e-14-5.64e-12Natural Gas Prod.Nickel empdssecondary-2.82e-14-1.25e-11CRT IncinerationHalogenated matter (organic)secondary-3.67e-14-1.63e-11CRT IncinerationCyanide (-1)secondary-2.17e-13-9.64e-11Natural Gas Prod.Lead empdssecondary-2.33e-13-1.04e-10CRT IncinerationNickel empdssecondary-2.3e-13-1.04e-10CRT IncinerationNickel empdssecondary-5.72e-13-2.54e-10Natural Gas Prod.Lead empdssecondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium empdssecondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium empdssecondary-1.74e-12-6.41e-10Natural Gas Prod.Cadmium empdssecondary-1.74e-12-6.41e-10Natural Gas Prod.Barium empdssecondary-1.74e-12-6.94e-10Natural Gas Prod.Cadmium empdssecondary-1.44e-12-6.41e-10Natural Gas	LPG Production	Nickel cmpds	secondary	2.40e-14	1.07e-11
LPG ProductionCyanide (-1)secondary7.05e-153.14e-12LPG ProductionHalogenated matter (organic)secondary2.01e-158.96e-13Natural Gas Prod.Halogenated matter (organic)secondary-2.37e-15-1.05e-12Natural Gas Prod.Cyanide (-1)secondary-8.28e-15-3.68e-12Natural Gas Prod.Mercury compoundssecondary-1.27e-14-5.64e-12Natural Gas Prod.Nickel empdssecondary-2.82e-14-1.25e-11CRT IncinerationHalogenated matter (organic)secondary-3.67e-14-1.63e-11CRT IncinerationCyanide (-1)secondary-2.17e-13-9.64e-11Natural Gas Prod.Lead cmpdssecondary-2.33e-13-1.04e-10CRT IncinerationNickel empdssecondary-2.33e-13-1.04e-10CRT IncinerationNickel cmpdssecondary-2.33e-13-1.04e-10Ratural Gas Prod.Lead cmpdssecondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium cmpdssecondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium cmpdssecondary-1.44e-12-6.41e-10Natural Gas Prod.Barium cmpdssecondary-1.77e-12-7.89e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.77e-12-7.89e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.77e-12-7.89e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.27e-12-8.91e-10 <td>LPG Production</td> <td>Mercury compounds</td> <td>secondary</td> <td>1.08e-14</td> <td>4.80e-12</td>	LPG Production	Mercury compounds	secondary	1.08e-14	4.80e-12
LPG ProductionHalogenated matter (organic)secondary2.01e-158.96e-13Natural Gas Prod.Halogenated matter (organic)secondary-2.37e-15-1.05e-12Natural Gas Prod.Cyanide (-1)secondary-8.28e-15-3.68e-12Natural Gas Prod.Mercury compoundssecondary-1.27e-14-5.64e-12Natural Gas Prod.Nickel cmpdssecondary-2.82e-14-1.25e-11CRT IncinerationHalogenated matter (organic)secondary-3.67e-14-1.63e-11CRT IncinerationCyanide (-1)secondary-2.17e-13-9.64e-11Natural Gas Prod.Toluenesecondary-2.33e-13-1.04e-10CRT IncinerationNickel cmpdssecondary-2.33e-13-1.04e-10Natural Gas Prod.Lead cmpdssecondary-2.38e-13-1.05e-10Natural Gas Prod.Lead cmpdssecondary-5.72e-13-2.54e-10Natural Gas Prod.Halogenated matter (organic)secondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium cmpdssecondary-1.77e-12-7.89e-10Natural Gas Prod.Barium cmpdssecondary-1.77e-12-7.89e-10Natural Gas Prod.Cyanide (-1)secondary-1.77e-12-7.89e-10Natural Gas Prod.Cyanide (-1)secondary-1.77e-12-7.89e-10Natural Gas Prod.Cyanide (-1)secondary-1.77e-12-7.89e-10Natural Gas Prod.Cyanide (-1)secondary-1.77e-12-7.89e-10<	LPG Production	Cyanide (-1)	secondary	7.05e-15	3.14e-12
Natural Gas Prod.Halogenated matter (organic)secondary-2.37e-15-1.05e-12Natural Gas Prod.Cyanide (-1)secondary-8.28e-15-3.68e-12Natural Gas Prod.Mercury compoundssecondary-1.27e-14-5.64e-12Natural Gas Prod.Nickel empdssecondary-2.82e-14-1.25e-11CRT IncinerationHalogenated matter (organic)secondary-3.67e-14-1.63e-11CRT IncinerationCyanide (-1)secondary-2.17e-13-9.64e-11Natural Gas Prod.Toluenesecondary-2.33e-13-1.04e-10CRT IncinerationLead empdssecondary-2.33e-13-1.04e-10CRT IncinerationNickel empdssecondary-2.32e-13-1.04e-10Natural Gas Prod.Lead empdssecondary-2.37e-13-9.64e-11Natural Gas Prod.Lead empdssecondary-2.32e-13-1.04e-10CRT IncinerationNickel empdssecondary-2.32e-13-1.04e-10Fuel Oil #4 Prod.Halogenated matter (organic)secondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium empdssecondary-6.98e-13-3.10e-10Natural Gas Prod.Barium empdssecondary-1.77e-12-7.89e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.77e-12-7.89e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.77e-12-7.89e-10Fuel Oil #4 Prod.Aromatic hydrocarbonssecondary-2.00e-12-8.91e-10 <td>LPG Production</td> <td>Halogenated matter (organic)</td> <td>secondary</td> <td>2.01e-15</td> <td>8.96e-13</td>	LPG Production	Halogenated matter (organic)	secondary	2.01e-15	8.96e-13
Natural Gas Prod.Cyanide (-1)secondary-8.28e-15-3.68e-12Natural Gas Prod.Mercury compoundssecondary-1.27e-14-5.64e-12Natural Gas Prod.Nickel cmpdssecondary-2.82e-14-1.25e-11CRT IncinerationHalogenated matter (organic)secondary-3.67e-14-1.63e-11CRT IncinerationCyanide (-1)secondary-2.17e-13-5.71e-11Natural Gas Prod.Toluenesecondary-2.33e-13-1.04e-10CRT IncinerationNickel cmpdssecondary-2.33e-13-1.04e-10Natural Gas Prod.Lead cmpdssecondary-5.72e-13-2.54e-10CRT IncinerationNickel cmpdssecondary-5.72e-13-2.54e-10Natural Gas Prod.Halogenated matter (organic)secondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium cmpdssecondary-6.98e-13-3.10e-10Natural Gas Prod.Cadmium cmpdssecondary-1.24e-12-6.41e-10Natural Gas Prod.Barium cmpdssecondary-1.77e-12-7.89e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.77e-12-7.89e-10Fuel Oil #4 Prod.Cyanide (-1)secondary-1.77e-12-7.89e-10Fuel Oil #4 Prod.Cyanide (-1)secondary-1.77e-12-7.89e-10Fuel Oil #4 Prod.Cyanide (-1)secondary-2.00e-12-8.91e-10Fuel Oil #4 Prod.Mercury compoundssecondary-3.06e-12-1.36e-00Fuel Oil	Natural Gas Prod.	Halogenated matter (organic)	secondary	-2.37e-15	-1.05e-12
Natural Gas Prod.Mercury compoundssecondary-1.27e-14-5.64e-12Natural Gas Prod.Nickel cmpdssecondary-2.82e-14-1.25e-11CRT IncinerationHalogenated matter (organic)secondary-3.67e-14-1.63e-11CRT IncinerationCyanide (-1)secondary-1.29e-13-5.71e-11Natural Gas Prod.Toluenesecondary-2.17e-13-9.64e-11Natural Gas Prod.Lead cmpdssecondary-2.33e-13-1.04e-10CRT IncinerationNickel cmpdssecondary-4.38e-13-1.95e-10Natural Gas Prod.Lead cmpdssecondary-5.72e-13-2.54e-10Ruel Oil #4 Prod.Halogenated matter (organic)secondary-6.98e-13-3.10e-10Natural Gas Prod.Cadmium cmpdssecondary-1.44e-12-6.41e-10Natural Gas Prod.Barium cmpdssecondary-1.77e-12-7.89e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.77e-12-7.89e-10Fuel Oil #4 Prod.Cyanide (-1)secondary-1.36e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.30e-12Fuel Oil #4 Prod.Cyanide (-1)secondary-1.77e-12-7.89e-10Fuel Oil #4 Prod.Cyanide (-1)secondary-2.00e-12-8.91e-100Fuel Oil #4 Prod.Cyanide (-1)secondary-3.06e-12-1.36e-00Fuel Oil #4 Prod.Mercury compoundssecondary-3.06e-12-1.36e-00 <tr <td="">Fuel Oil #4 ProdMercury</tr>	Natural Gas Prod.	Cyanide (-1)	secondary	-8.28e-15	-3.68e-12
Natural Gas Prod.Nickel cmpdssecondary-2.82e-14-1.25e-11CRT IncinerationHalogenated matter (organic)secondary-3.67e-14-1.63e-11CRT IncinerationCyanide (-1)secondary-1.29e-13-5.71e-11Natural Gas Prod.Toluenesecondary-2.17e-13-9.64e-11Natural Gas Prod.Lead cmpdssecondary-2.33e-13-1.04e-10CRT IncinerationNickel cmpdssecondary-2.32e-13-1.04e-10CRT IncinerationNickel cmpdssecondary-5.72e-13-2.54e-10Natural Gas Prod.Halogenated matter (organic)secondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium cmpdssecondary-6.98e-13-3.10e-10Natural Gas Prod.Barium cmpdssecondary-1.44e-12-6.41e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.77e-12-7.89e-10Fuel Oil #4 Prod.Cyanide (-1)secondary-1.30e-12-8.91e-100Fuel Oil #4 Prod.Cyanide (-1)secondary-1.77e-12-7.89e-100Fuel Oil #4 Prod.Cyanide (-1)secondary-2.00e-12-8.91e-100Fuel Oil #4 Prod.Mercury compoundssecondary-3.06e-12-1.36e.000	Natural Gas Prod.	Mercury compounds	secondary	-1.27e-14	-5.64e-12
CRT IncinerationHalogenated matter (organic)secondary-3.67e-14-1.63e-11CRT IncinerationCyanide (-1)secondary-1.29e-13-5.71e-11Natural Gas Prod.Toluenesecondary-2.17e-13-9.64e-11Natural Gas Prod.Lead cmpdssecondary-2.33e-13-1.04e-10CRT IncinerationNickel cmpdssecondary-4.38e-13-1.95e-10Fuel Oil #4 Prod.Halogenated matter (organic)secondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium cmpdssecondary-6.98e-13-3.10e-10Natural Gas Prod.Barium cmpdssecondary-1.44e-12-6.41e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.77e-12-7.89e-10Fuel Oil #4 Prod.Cyanide (-1)secondary-1.30e-12-8.91e-100Fuel Oil #4 Prod.Cyanide (-1)secondary-2.00e-12-8.91e-100Fuel Oil #4 Prod.Mercury compoundssecondary-2.00e-12-1.36e.000	Natural Gas Prod.	Nickel cmpds	secondary	-2.82e-14	-1.25e-11
CRT IncinerationCyanide (-1)secondary-1.29e-13-5.71e-11Natural Gas Prod.Toluenesecondary-2.17e-13-9.64e-11Natural Gas Prod.Lead cmpdssecondary-2.33e-13-1.04e-10CRT IncinerationNickel cmpdssecondary-4.38e-13-1.95e-10Fuel Oil #4 Prod.Halogenated matter (organic)secondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium cmpdssecondary-6.98e-13-3.10e-10Natural Gas Prod.Barium cmpdssecondary-1.44e-12-6.41e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.77e-12-7.89e-10Fuel Oil #4 Prod.Cyanide (-1)secondary-2.00e-12-8.91e-100Fuel Oil #4 Prod.Mercury compoundssecondary-3.06e-12-1.36e.000	CRT Incineration	Halogenated matter (organic)	secondary	-3.67e-14	-1.63e-11
Natural Gas Prod.Toluenesecondary-2.17e-13-9.64e-11Natural Gas Prod.Lead cmpdssecondary-2.33e-13-1.04e-10CRT IncinerationNickel cmpdssecondary-4.38e-13-1.95e-10Fuel Oil #4 Prod.Halogenated matter (organic)secondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium cmpdssecondary-6.98e-13-3.10e-10Natural Gas Prod.Barium cmpdssecondary-1.44e-12-6.41e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.77e-12-7.89e-10Fuel Oil #4 Prod.Cyanide (-1)secondary-2.00e-12-8.91e-10Fuel Oil #4 Prod.Mercury compoundssecondary-3.06e-13-1.36e-00	CRT Incineration	Cyanide (-1)	secondary	-1.29e-13	-5.71e-11
Natural Gas Prod.Lead cmpdssecondary-2.33e-13-1.04e-10CRT IncinerationNickel cmpdssecondary-4.38e-13-1.95e-10Fuel Oil #4 Prod.Halogenated matter (organic)secondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium cmpdssecondary-6.98e-13-3.10e-10Natural Gas Prod.Barium cmpdssecondary-1.44e-12-6.41e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.77e-12-7.89e-10Fuel Oil #4 Prod.Cyanide (-1)secondary-2.00e-12-8.91e-100Fuel Oil #4 Prod.Mercury compoundssecondary-3.06e-12-1.36e.000	Natural Gas Prod.	Toluene	secondary	-2.17e-13	-9.64e-11
CRT IncinerationNickel mpdssecondary-4.38e-13-1.95e-10Fuel Oil #4 Prod.Halogenated matter (organic)secondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium empdssecondary-6.98e-13-3.10e-10Natural Gas Prod.Barium empdssecondary-1.44e-12-6.41e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.77e-12-7.89e-10Fuel Oil #4 Prod.Cyanide (-1)secondary-2.00e-12-8.91e-100Fuel Oil #4 Prod.Mercury compoundssecondary-3.06e-12-1.36e-000	Natural Gas Prod.	Lead cmpds	secondary	-2.33e-13	-1.04e-10
Fuel Oil #4 Prod.Halogenated matter (organic)secondary-5.72e-13-2.54e-10Natural Gas Prod.Cadmium empdssecondary-6.98e-13-3.10e-10Natural Gas Prod.Barium empdssecondary-1.44e-12-6.41e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.77e-12-7.89e-10Fuel Oil #4 Prod.Cyanide (-1)secondary-2.00e-12-8.91e-100Fuel Oil #4 ProdMercury compoundssecondary-3.06e-12-1.36e-000	CRT Incineration	Nickel cmpds	secondary	-4.38e-13	-1.95e-10
Natural Gas Prod.Cadmium cmpdssecondary-6.98e-13-3.10e-10Natural Gas Prod.Barium cmpdssecondary-1.44e-12-6.41e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.77e-12-7.89e-10Fuel Oil #4 Prod.Cyanide (-1)secondary-2.00e-12-8.91e-100Fuel Oil #4 ProdMercury compoundssecondary-3.06e-12-1.36e-000	Fuel Oil #4 Prod.	Halogenated matter (organic)	secondary	-5.72e-13	-2.54e-10
Natural Gas Prod.Barium cmpdssecondary-1.44e-12-6.41e-10Natural Gas Prod.Aromatic hydrocarbonssecondary-1.77e-12-7.89e-10Fuel Oil #4 Prod.Cyanide (-1)secondary-2.00e-12-8.91e-10Fuel Oil #4 Prod.Mercury compoundssecondary-3.06e-12-1.36e-00	Natural Gas Prod.	Cadmium cmpds	secondary	-6.98e-13	-3.10e-10
Natural Gas Prod.Aromatic hydrocarbonssecondary-1.77e-12-7.89e-10Fuel Oil #4 Prod.Cyanide (-1)secondary-2.00e-12-8.91e-10Fuel Oil #4 ProdMercury compoundssecondary-3.06e-12-1.36e-00	Natural Gas Prod.	Barium cmpds	secondary	-1.44e-12	-6.41e-10
Fuel Oil #4 Prod.Cyanide (-1)secondary-2.00e-12-8.91e-10Fuel Oil #4 ProdMercury compoundssecondary-3.06e-12-1.36e-09	Natural Gas Prod.	Aromatic hydrocarbons	secondary	-1.77e-12	-7.89e-10
Fuel Oil #4 Prod Mercury compounds secondary -3.06e-12 -1.36e-00	Fuel Oil #4 Prod.	Cyanide (-1)	secondary	-2.00e-12	-8.91e-10
	Fuel Oil #4 Prod.	Mercury compounds	secondary	-3.06e-12	-1.36e-09
Fuel Oil #4 Prod. Nickel cmpds secondary -6.82e-12 -3.03e-09	Fuel Oil #4 Prod.	Nickel cmpds	secondary	-6.82e-12	-3.03e-09
CRT Incineration Aromatic hydrocarbons secondary -2.75e-11 -1.22e-08	CRT Incineration	Aromatic hydrocarbons	secondary	-2.75e-11	-1.22e-08
Fuel Oil #4 Prod. Toluene secondary -5.24e-11 -2.33e-08	Fuel Oil #4 Prod.	Toluene	secondary	-5.24e-11	-2.33e-08
Fuel Oil #4 Prod. Lead cmpds secondary -5.64e-11 -2.51e-08	Fuel Oil #4 Prod.	Lead cmpds	secondary	-5.64e-11	-2.51e-08
Fuel Oil #4 Prod.Chromium (III)secondary-1.03e-10-4.59e-08	Fuel Oil #4 Prod.	Chromium (III)	secondary	-1.03e-10	-4.59e-08
Fuel Oil #4 Prod. Chromium (VI) secondary -1.44e-10 -6.40e-08	Fuel Oil #4 Prod.	Chromium (VI)	secondary	-1.44e-10	-6.40e-08
Natural Gas Prod. Copper (+1 & +2) secondary -1.57e-10 -6.98e-08	Natural Gas Prod.	Copper (+1 & +2)	secondary	-1.57e-10	-6.98e-08
Fuel Oil #4 Prod. Cadmium cmpds secondary -1.69e-10 -7.51e-08	Fuel Oil #4 Prod.	Cadmium cmpds	secondary	-1.69e-10	-7.51e-08
Fuel Oil #4 Prod. Barium cmpds secondary -3.48e-10 -1.55e-07	Fuel Oil #4 Prod.	Barium cmpds	secondary	-3.48e-10	-1.55e-07
Fuel Oil #4 Prod. Aromatic hydrocarbons secondary -4.29e-10 -1.91e-07	Fuel Oil #4 Prod.	Aromatic hydrocarbons	secondary	-4.29e-10	-1.91e-07
Natural Gas Prod. Chromium (III) secondary -1.10e-09 -4.90e-07	Natural Gas Prod.	Chromium (III)	secondary	-1.10e-09	-4.90e-07
Natural Gas Prod. Chromium (VI) secondary -1.54e-09 -6.83e-07	Natural Gas Prod.	Chromium (VI)	secondary	-1.54e-09	-6.83e-07
CRT Incineration Copper (+1 & +2) secondary -2.44e-09 -1.08e-06	CRT Incineration	Copper (+1 & +2)	secondary	-2.44e-09	-1.08e-06
CRT Incineration Zinc (+2) secondary -8.28e-09 -3.68e-06	CRT Incineration	Zinc (+2)	secondary	-8.28e-09	-3.68e-06
Fuel Oil #4 Prod. Zinc (+2) secondary -1.19e-08 -5.27e-06	Fuel Oil #4 Prod.	Zinc (+2)	secondary	-1.19e-08	-5.27e-06
Natural Gas Prod. Nitrate secondary -1.51e-08 -6.71e-06	Natural Gas Prod.	Nitrate	secondary	-1.51e-08	-6.71e-06
Natural Gas Prod. Aluminum (+3) secondary -2.49e-08 -1.11e-05	Natural Gas Prod.	Aluminum (+3)	secondary	-2.49e-08	-1.11e-05
Fuel Oil #4 Prod. Copper (+1 & +2) secondary -3.80e-08 -1.69e-05	Fuel Oil #4 Prod.	Copper (+1 & +2)	secondary	-3.80e-08	-1.69e-05
Natural Gas Prod. Phenol secondary -4.34e-08 -1.93e-05	Natural Gas Prod.	Phenol	secondary	-4.34e-08	-1.93e-05
Natural Gas Prod. Fluorides (F-) secondary -5.98e-08 -2.66e-05	Natural Gas Prod.	Fluorides (F-)	secondary	-5.98e-08	-2.66e-05
Natural Gas Prod. Zinc (+2) secondary -9.71e-08 -4.32e-05	Natural Gas Prod.	Zinc (+2)	secondary	-9.71e-08	-4.32e-05
Fuel Oil #4 Prod. Fluorides (F-) secondary -2.45e-07 -1.09e-04	Fuel Oil #4 Prod.	Fluorides (F-)	secondary	-2.45e-07	-1.09e-04
Fuel Oil #4 Prod.Nitratesecondary-3.15e-07-1.40e-04	Fuel Oil #4 Prod.	Nitrate	secondary	-3.15e-07	-1.40e-04
CRT Incineration Aluminum (+3) secondary -3.87e-07 -1.72e-04	CRT Incineration	Aluminum (+3)	secondary	-3.87e-07	-1.72e-04
CRT Incineration Phenol secondary -6.69e-07 -2.97e-04	CRT Incineration	Phenol	secondary	-6.69e-07	-2.97e-04
CRT Incineration Nitrate secondary -3.60e-06 -1.60e-03	CRT Incineration	Nitrate	secondary	-3.60e-06	-1.60e-03
Fuel Oil #4 Prod. Aluminum (+3) secondary -6.03e-06 -2.68e-03	Fuel Oil #4 Prod.	Aluminum (+3)	secondary	-6.03e-06	-2.68e-03

Table M-37. CRT LCIA Results for the Aquatic Toxicity Impact Category

Process Group	Material	LCI Data Type	Aquatic Toxicity (tox-kg)	% of Total
Fuel Oil #4 Prod.	Phenol	secondary	-1.04e-05	-4.63e-03
CRT Incineration	Fluorides (F-)	secondary	-1.51e-05	-6.73e-03
CRT Incineration	Ammonia	secondary	-3.00e-04	-1.34e-01
Total End-of-life			-2.50e-04	-1.11e-01
Total All Life-cycle Stages			2.25e-01	1.00e+02

			Aquatic Toxicity	
Process Group	Material	LCI Data Type	(tox-kg)	% of Total
Materials Processing Life-cycle	Stage			
PMMA Sheet Prod.	Ammonia	secondary	3.29e-02	6.34e-01
Aluminum Prod.	Aluminum (+3)	secondary	1.03e-02	1.98e-01
Aluminum Prod.	Copper (+1 & +2)	secondary	7.92e-03	1.53e-01
Aluminum Prod.	Zinc (+2)	secondary	2.40e-03	4.62e-02
Steel Prod., cold-rolled, semi-finished	Phosphorus (yellow or white)	secondary	2.25e-03	4.34e-02
Steel Prod., cold-rolled, semi-finished	Ammonia	secondary	1.37e-03	2.64e-02
Polycarbonate Production	Copper (+1 & +2)	secondary	6.84e-04	1.32e-02
PMMA Sheet Prod.	Copper (+1 & +2)	secondary	5.09e-04	9.81e-03
Steel Prod., cold-rolled, semi-finished	Copper (+1 & +2)	secondary	4.98e-04	9.60e-03
Aluminum Prod.	Barium sulfate	secondary	3.38e-04	6.51e-03
PET Resin Production	Copper (+1 & +2)	secondary	2.41e-04	4.64e-03
Polycarbonate Production	Mercury compounds	secondary	2.40e-04	4.63e-03
Styrene-butadiene Copolymer Prod.	Copper (+1 & +2)	secondary	2.40e-04	4.63e-03
PMMA Sheet Prod.	Mercury compounds	secondary	1.79e-04	3.44e-03
Aluminum Prod.	Titanium tetrachloride	secondary	1.70e-04	3.28e-03
Steel Prod., cold-rolled, semi-finished	Aluminum (+3)	secondary	1.34e-04	2.58e-03
Aluminum Prod.	Strontium (Sr II)	secondary	1.19e-04	2.29e-03
Steel Prod., cold-rolled, semi-finished	Nitrogen dioxide	secondary	9.68e-05	1.87e-03
Steel Prod., cold-rolled, semi-finished	$Z_{inc}(+2)$	secondary	9.60e-05	1.85e-03
Polycarbonate Production	$Z_{\text{inc}}(+2)$	secondary	9 53e-05	1 84e-03
Steel Prod cold-rolled semi-finished	Fluorides (F-)	secondary	9.44e-05	1.82e-03
Styrene-butadiene Conolymer Prod	Mercury compounds	secondary	8 42e-05	1.62e-03
Steel Prod cold-rolled semi-finished	Strontium (Sr II)	secondary	7 31e-05	1.41e-03
PMMA Sheet Prod	$Z_{inc}(+2)$	secondary	7.08e-05	1 37e-03
Polycarbonate Production	Chlorine	secondary	6.81e-05	1.31e-03
Polycarbonate Production	Ammonia	secondary	6.63e-05	1.28e-03
Aluminum Prod	Lead cmpds	secondary	6.40e-05	1.23e-03
Aluminum Prod	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI) secondary	5.76e-05	1.25e 03
PMMA Sheet Prod	Chlorine	secondary	5.06e-05	9 77e-04
Aluminum Prod	Cadmium empds	secondary	4 32e 05	8 3/e 0/
Steel Prod cold-rolled semi-finished	Mercury compounds	secondary	3.92e-05	7 56e-04
Aluminum Prod	Barium empde	secondary	3.48e.05	6 72e 04
Aluminum Prod	Nitrata	secondary	2 400 05	6 550 04
Staal Brad, and rolled somi finished	Parium sulfata	secondary	3.406-03	6.400.04
DET Pasin Production	Zing (+2)	secondary	2 350 05	6.470.04
PET Resili Production	Zinc (+2)	secondary	3.556-05	6.476-04
Styrene-butadiene Copolymer Prod.	Zilic (+2) Vulana (miyad isamara)	secondary	3.546-05	6.152.04
Delyaerhamete Dreduction	Dhanal	secondary	2.102.05	5.0% 0.04
DET Design Dre destion	Chloring	secondary	3.10e-05	3.986-04
PET Resin Production	Chlorine	secondary	2.40e-05	4.62e-04
Styrene-butadiene Copolymer Prod.	Chiorine N. 1	secondary	2.396-05	4.01e-04
		secondary	1.42e-05	2.74e-04
Steel Prod., cold-rolled, semi-finished	Silver compounds	secondary	1.39e-05	2.68e-04
INALUFAI GAS Prod.	$\operatorname{Zinc}(+2)$	secondary	1.35e-05	2.61e-04
Aluminum Prod.	Chromium (III)	secondary	1.19e-05	2.30e-04
Aluminum Prod.	Aromatic hydrocarbons	secondary	1.05e-05	2.03e-04
PMMA Sheet Prod.	Phenol	secondary	1.04e-05	2.00e-04
Steel Prod., cold-rolled, semi-finished	Lead cmpds	secondary	8.52e-06	1.64e-04
Aluminum Prod.	Manganese cmpds	secondary	8.48e-06	1.64e-04

Process Group Material ICI Data Type (tockg) % of Total Nutral Gis Prod. Fluoride (r) secondary 5.32-66 1.61-64 PMA Sheet Prod. Quride (-1) secondary 7.60-66 1.47-64 Aluminum Prod. Fluoride (-1) secondary 6.37-606 1.31-640 Steel Prod., cold-rolled, semi-finished Aromatic hydrocarbons secondary 6.37-606 1.21-644 Aluminum Prod. Vandam (3:5, '5:7) secondary 6.08-66 1.17-644 Aluminum Prod. Selerinin secondary 5.10e-66 9.68-045 Aluminum Prod. Selerinin secondary 5.10e-66 6.98-045 Aluminum Prod. Xylere (mixed isomers) secondary 3.47-06 6.69-05 Natural Gias Prod. Aluminum (+3) secondary 3.47-06 6.89-05 Stater Good, Cold-cold, semi-finished Arserio: mphs secondary 2.53-06 4.88-05 Aluminum Prod. Arserio: mphs secondary 2.14-06 6.89-05 Steel Prod., cold-cold-cold, semi-finis				Aquatic Toxicity	
Natural Gas Prod. Fluorides (F-) secondary 8.38-045 1.616-04 Steel Prod., cold-rolled, semi-finished Burinum empts secondary 7.67e-06 1.47e-04 Steel Prod., cold-rolled, semi-finished Anuminum (T-) secondary 6.37e-06 1.21e-04 Steel Prod., cold-rolled, semi-finished Anuminum (T-) secondary 6.37e-06 1.21e-04 Anuminum Prod. Selarium secondary 6.06e-06 1.17e-04 Natural Gas Prod. Phenol secondary 5.90e-06 1.14e-04 Steel Prod., cold-rolled, semi-finished Tin (Sr++, Sr++) secondary 3.74e-06 6.63e-05 Natural Gas Prod. Aluminum (T-3) secondary 3.74e-06 6.63e-05 Natural Gas Prod. Aluminum (T-3) secondary 3.74e-06 6.642-045 Aluminum Prod. Argenci empts secondary 2.73e-06 4.88e-05 Aluminum Prod. Reserva secondary 2.73e-06 4.88e-05 Natural Gas Prod. Nitrates'initres secondary 1.74e-06 3.35e-05 <	Process Group	Material	LCI Data Type	(tox-kg)	% of Total
PMMA Sheat Prod. Cyanids (-1) secondary 7.76/2-66 1.48-64 Aluminum Prod. Barium cropds secondary 7.76/2-66 1.17-64 Aluminum Prod. Horoids secondary 6.78-66 1.21-64 Aluminum Prod. Vanadium (Y3+, V5+) secondary 6.88-66 1.17-64 Aluminum Prod. Vanadium (Y3+, V5+) secondary 5.08-66 1.17-64 Aluminum Prod. Sclerium secondary 5.108-66 9.88-645 Aluminum Prod. Sclerium secondary 5.108-66 6.88-645 Aluminum Prod. Xytee (mixed isomers) secondary 3.12-66 6.42-645 Aluminum Prod. Aluminum (+3) secondary 2.32-66 4.28-65 Syteme-buildene Copolymer Prod. Nitrate secondary 2.32-66 4.88-645 Syteme-buildene Copolymer Prod. Nitrate secondary 2.32-66 4.28-65 Steel Prod., cold-colid, semi-finished Phenol secondary 1.74-66 6.35-605 Steel Prod., cold-colid, semi-finished Phen	Natural Gas Prod.	Fluorides (F-)	secondary	8.33e-06	1.61e-04
Steel Prod., cold-rolled, semi-finished Fluoride secondary 6.78-06 1.31-644 Steel Prod., cold-rolled, semi-finished Atomatic hydrocarbons secondary 6.28-06 1.31-644 Aluminum Prod. Vanadium (*3) secondary 6.28-06 1.31-644 Aluminum Prod. Vanadium (*3) secondary 5.06-06 1.11-644 Aluminum Prod. Steel Irod, cold-rolled, semi-finished Tin (Sh=*, Sh+*) secondary 5.00-06 9.81-05 Aluminum Prod. Xyleer (iniced isomeres) secondary 4.70-06 9.06-05 Polycarbonate Production Aluminum (~3) secondary 3.47-06 6.63-05 Matural Case Prod. Aluminum (~3) secondary 3.33-06 6.42-05 Aluminum Prod. Alernicum (~3) secondary 2.33-06 4.88-05 Steel Prod. Nitrate Strate secondary 2.33-06 4.88-05 Matura Case Prod. Nitrate Strate secondary 1.74-06 3.35-05 Steel Prod., cold-rolled, semi-finished Phonol secondary 1.48-06 <	PMMA Sheet Prod.	Cyanide (-1)	secondary	7.67e-06	1.48e-04
Aluminum Prod. Fluoride secondary 6.78e-06 1.31e/44 Ster Pord., dol-Folds, smi-finishe Atominum (+3) secondary 6.08e-06 1.121e/44 Natural Gas Prod. Vanadium (+3-, X5-) secondary 6.08e-06 1.17e-04 Natural Gas Prod. Phenol secondary 5.00e-06 1.17e-04 Aluminum Prod. Selerium secondary 5.00e-06 0.81e-05 Aluminum Prod. Kglere (mixed isomers) secondary 4.88e-06 8.61e-06 Natural Gas Prod. Aluminum (+3) secondary 3.19e-06 6.68e-05 Natural Gas Prod. Aluminum (+3) secondary 3.23e-06 6.68e-05 Stere Prod., cold-rolled, semi-finished Arrenic cmpds secondary 2.32e-06 4.88e-05 Natural Gas Prod. Nirate secondary 2.32e-06 4.88e-05 Stere Prod., cold-rolled, semi-finished Brenzene secondary 1.28e-06 3.04e-05 Stere Prod., cold-rolled, semi-finished Phenol secondary 1.28e-06 3.04e-05 S	Steel Prod., cold-rolled, semi-finished	Barium cmpds	secondary	7.60e-06	1.47e-04
Steel Prod., cold-rolled, semi-finished Atominum (*3) secondary 6.28-06 1.12-04 Atuminum Prod. Vanadium (*3) secondary 6.08-06 1.17-04 Mutural Gas Prod. Phenol secondary 6.08-06 1.17-04 Atuminum Prod. Selerinim secondary 5.09-06 1.17-04 Steel Prod., cold-rolled, semi-finished Tin (Sh+*, Sh+*) secondary 4.70-06 9.08-05 Polycarbonate Production Aluminum (*3) secondary 3.47-06 6.63-05 Natural Cas Prod. Aluminum (*3) secondary 3.33-06 6.12-05 System-bundine Coplymer Prod. Nitrate secondary 2.33-06 4.88-05 Muninum Prod. Rescruce secondary 2.33-06 4.88-05 Steel Prod. Nitrate secondary 2.23-06 4.88-05 Steel Prod. Nitrate sintifities secondary 2.20-06 4.38-06 Steel Prod., cold-rolled, semi-finished Benzene secondary 1.74-06 3.35-06 Steel Prod., cold-rolled, semi-finished </td <td>Aluminum Prod.</td> <td>Fluoride</td> <td>secondary</td> <td>6.78e-06</td> <td>1.31e-04</td>	Aluminum Prod.	Fluoride	secondary	6.78e-06	1.31e-04
Styrens-buildene Copolymer Prod. Aluminum (-3) secondary 6.08-06 1.17-04 Natural Gas Prod. Phenol secondary 5.00-66 1.17-04 Natural Gas Prod. Sclenium secondary 5.00-66 1.17-04 Aluminum Prod. Sclenium secondary 5.00-66 9.84-05 Aluminum Prod. Kylene (mixed isomers) secondary 4.88-06 8.63-065 Natural Gas Prod. Aluminum (-3) secondary 3.32-06 6.63-05 Natural Gas Prod. Aluminum (-3) secondary 3.32-06 6.83-05 Aluminum Prod. Arsenic empds secondary 2.32-06 4.88-05 Aluminum Prod. Benzene secondary 2.32-06 4.88-05 Natural Gas Prod. Nitrate secondary 2.16-06 4.83-05 Steel Prod. Nitrate secondary 2.16-06 4.33-05 Steel Prod. Nitrate secondary 1.74-06 3.33-05 Steel Prod. Nitrate secondary 1.74-06 3.34-05	Steel Prod., cold-rolled, semi-finished	Aromatic hydrocarbons	secondary	6.37e-06	1.23e-04
Aluminum Prod. Vandium (V3+, V5+) secondary 6.06-06 1.17e-04 Munium Serod. Phenol secondary 5.90e-06 1.14e-04 Aluminum Prod. Sclenium secondary 5.90e-06 1.14e-04 Steel Prod., cold-rolled, semi-finished Tin (Sn++, Sn+) secondary 4.70e-06 9.06e-05 Polyachonate Production Aluminum (+3) secondary 4.74e-06 6.06e-05 PMAA Sheet Prod. Aluminum (+3) secondary 3.33e-06 6.42e-05 Aluminum Prod. Asenic empds secondary 2.35e-06 4.88e-05 Sycrae-bualdienc Copolymer Prod. Nitrate secondary 2.35e-06 4.88e-05 Steel Prod. Nitrate-Sintritis secondary 2.10e-06 4.05e-05 Steel Prod. Nitrate-Sintritis secondary 1.74e-06 3.35e-05 Steel Prod. Nitrate-Sintritis secondary 1.74e-06 3.35e-05 Steel Prod. Nitrate-Sintritis secondary 1.74e-06 3.35e-05 Steel Prod. Colorine	Styrene-butadiene Copolymer Prod.	Aluminum (+3)	secondary	6.28e-06	1.21e-04
Natural Gas Prod. Phenol secondary 6.04e-06 1.17e-04 Stecl Prod., cold-rolled, semi-finished Tin (Sn++, Sn+) secondary 5.10e-06 9.84e-05 Aluminum Prod. Xylene (mixed isomers) secondary 4.70e-06 9.84e-05 Natural Gas Prod. Aluminum (+3) secondary 3.37e-06 6.67e-05 Natural Gas Prod. Aluminum (+3) secondary 3.37e-06 6.672-05 Aluminum Prod. Arsenic empds secondary 2.33e-06 6.15e-05 Stepter Prod. Nitrate secondary 2.35e-06 4.88e-05 Natural Gas Prod. Berzene secondary 2.12e-06 4.28e-05 Natural Gas Prod. Nitrate-secondary 1.74e-06 3.35e-05 58eel Prod., cold-rolled, seni-finished Phenol secondary 1.74e-06 3.35e-05 Steel Prod., cold-rolled, seni-finished Phenol secondary 1.74e-06 3.35e-05 Steel Prod., cold-rolled, seni-finished Phenol secondary 1.38e-06 3.04e-05 Steel Prod., cold-rolled, seni-finished	Aluminum Prod.	Vanadium (V3+, V5+)	secondary	6.06e-06	1.17e-04
Aluminum Prod. Selenium secondary 5.00-406 1.146-04 Aluminum Prod. Xylene (mixed isomers) secondary 4.70e-06 9.08e-05 Polycarbonate Production Aluminum (13) secondary 4.48e-06 9.05e-05 PMA Sheel Prod. Aluminum (13) secondary 3.33e-06 6.63e-05 Syrene-butadiene Copolymer Prod. Nitrate: secondary 2.53e-06 4.88e-05 Aluminum Prod. Arsenic empds secondary 2.53e-06 4.88e-05 Syrene-butadiene Copolymer Prod. Nitrate: secondary 2.25e-06 4.88e-05 Steel Prod., cold-rolled, semi-finished Benzene secondary 2.10e-06 4.08e-05 Steel Prod., cold-rolled, semi-finished Benzene secondary 1.74e-06 3.35e-05 Steel Prod., cold-rolled, semi-finished Benzene secondary 1.74e-06 3.35e-05 Steel Prod., cold-rolled, semi-finished Benzene secondary 1.74e-06 2.35e-05 Steel Prod., cold-rolled, semi-finished Benzene secondary 1.74e-06	Natural Gas Prod.	Phenol	secondary	6.04e-06	1.17e-04
Steel Prod., cold-rolled, semi-finished Tm (Sn ++, Sn ++) secondary 5.10e-06 9.84e-05 Polycarbonate Production Aluminum (+3) secondary 4.70e-06 9.00e-05 Natural Gas Prod. Aluminum (+3) secondary 3.47e-06 6.63e-05 Natural Gas Prod. Aluminum (+3) secondary 3.19e-06 6.15e-05 Styrene-butadiene Copolymer Prod. Nirate secondary 2.53e-06 4.88e-05 Muninum Prod. Benzene secondary 2.23e-06 4.88e-05 Steel Prod. Nirate Site Secondary 2.16e-06 4.88e-05 Steel Prod., cold-rolled, semi-finished Phenol secondary 1.74e-06 3.35e-05 Steel Prod., cold-rolled, semi-finished Phenol secondary 1.74e-06 3.35e-05 Steel Prod., cold-rolled, semi-finished Phenol secondary 1.74e-06 3.36e-05 Steel Prod., cold-rolled, semi-finished Natural Gas Prod. Natura Gas Prod. 1.74e-06 2.47e-05 Steel Prod., cold-rolled, semi-finished Natural Gas Prod. Natural Gas Prod. 3.36e-0	Aluminum Prod.	Selenium	secondary	5.90e-06	1.14e-04
Aluminum Prod. Xylene (mixed isomers) secondary 4.70e-06 9.06e-05 Natural Gas Prod. Aluminum (+3) secondary 3.47e-06 6.63e-05 Natural Gas Prod. Aluminum (+3) secondary 3.37e-06 6.63e-05 Styrene-butadiene Copolymer Prod. Nitrate secondary 2.53e-06 4.88e-05 Aluminum Prod. Benzene secondary 2.53e-06 4.88e-05 Aluminum Prod. Benzene secondary 2.13e-06 4.88e-05 Steel Prod. Nitrate secondary 2.10e-06 4.0se-05 Steel Prod. Nitrate secondary 1.74e-06 3.35e-05 Steel Prod., cold-rolled, semi-finished Phenol secondary 1.74e-06 3.35e-05 Aluminum Prod. Phenol secondary 1.74e-06 3.35e-05 Steel Prod., cold-rolled, semi-finished Chlorine secondary 1.74e-06 2.72e-05 Steel Prod., cold-rolled, semi-finished Chuminum (+3) secondary 1.41e-06 2.72e-05 Steel Prod., cold-rolled, semi-finish	Steel Prod., cold-rolled, semi-finished	Tin (Sn++, Sn4+)	secondary	5.10e-06	9.84e-05
Polycarbonate Production Aluminum (+3) secondary 4.48-606 8.63-605 PMMA Sheet Prod. Aluminum (+3) secondary 3.13-e.06 6.42-05 Aluminum Prod. Arsenic cmpds secondary 3.19-e.06 6.15-e.05 Syrgene-butadienc Copolyme Prod. Nitrate secondary 2.53-e.06 4.88-e05 Aluminum Prod. Benzene secondary 2.22-e.06 4.88-e05 Syrgene-butadienc Copolyme Prod. Nitrates secondary 2.10-e.06 4.88-e05 Steel Prod., cold-rolled, semi-finished Phenol secondary 1.74-e06 3.35-e05 Steel Prod., cold-rolled, semi-finished Benzene secondary 1.58-e06 3.04-e05 Steel Prod., cold-rolled, semi-finished Chouren secondary 1.47e-06 2.32-e05 Steel Prod., cold-rolled, semi-finished Molybietrum (Mo II, Mo IV, Mo V, Mo VI) secondary 1.47e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Molybietrum (Mo II, Mo IV, Mo V, Mo VI) secondary 1.31e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Molybietrum (Mo II, Mo IV, Mo V, Mo	Aluminum Prod.	Xylene (mixed isomers)	secondary	4.70e-06	9.06e-05
Natural Gas Prod. Aluminum (+3) secondary 3.13e-06 6.642e-05 Aluminum Prod. Arsenic empds secondary 3.13e-06 6.42e-05 Aluminum Prod. Brazne secondary 2.53e-06 4.88e-05 Aluminum Prod. Brazne secondary 2.53e-06 4.88e-05 MAM Sheet Prod. Nitrates'nitrites secondary 2.12e-06 4.88e-05 Natural Gas Prod. Nitrate secondary 2.12e-06 4.88e-05 Steel Prod., ocld-rolled, semi-finished Phenol secondary 1.74e-06 3.35e-05 Steel Prod., ocld-rolled, semi-finished Benzene secondary 1.74e-06 3.35e-05 Steel Prod., ocld-rolled, semi-finished Chorine secondary 1.47e-06 3.40e-05 Steel Prod., ocld-rolled, semi-finished Chorine secondary 1.47e-06 2.27e-05 Steel Prod., ocld-rolled, semi-finished Chorine secondary 1.30e-06 2.27e-05 Steel Prod., ocld-rolled, semi-finished Molybelnum (Mo II, Mo IV, Mo V, Mo V, Mo V) Steel Prod., ocld-rolled, semi-finished	Polycarbonate Production	Aluminum (+3)	secondary	4.48e-06	8.63e-05
PMMA Sheel Prod. Aluminum (+3) secondary 3.38-06 6.42e-05 Aluminum Prod. Arsenic empds secondary 2.58-06 4.88e-05 Aluminum Prod. Benzene secondary 2.58-06 4.88e-05 Muntal Gas Prod. Nitrates' nitrates secondary 2.16e-06 4.05e-05 Steel Prod., cold-rolled, semi-finished Phenol secondary 1.74e-06 3.35e-05 Steel Prod., cold-rolled, semi-finished Phenol secondary 1.78e-06 3.35e-05 Steel Prod., cold-rolled, semi-finished Chlurne escondary 1.78e-06 3.36e-05 Aluminum Prod. Toluene secondary 1.78e-06 3.36e-05 Steel Prod., cold-rolled, semi-finished Chlorine secondary 1.47e-06 2.84e-05 Steel Prod., cold-rolled, semi-finished Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo V) secondary 1.38e-06 2.47e-05 Steel Prod., cold-rolled, semi-finished Cadmium cmpds secondary 1.31e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo V) secondary 1.18e	Natural Gas Prod.	Aluminum (+3)	secondary	3.47e-06	6.69e-05
Aluminum Prod. Arsenic empds secondary 31.9e-06 65.1e-05 Styrene-butadiene Copolymer Prod. Nitrate secondary 2.35e-06 4.88e-05 PMMA Sheet Prod. Nitrates/nitrites secondary 2.13e-06 4.88e-05 Steel Prod. Nitrates/nitrites secondary 2.10e-06 4.05e-05 Steel Prod., cold-rolled, semi-finished Phenol secondary 1.74e-06 3.35e-05 Aluminum Prod. Toluene secondary 1.78e-06 3.35e-05 Aluminum Prod. Toluene secondary 1.78e-06 3.04e-05 Steel Prod., cold-rolled, semi-finished Chlorine secondary 1.78e-06 3.04e-05 Steel Prod., cold-rolled, semi-finished Molyblenum (Mo II, Mo IV, Mo V, Mo VI) secondary 1.39e-06 2.69e-05 Steel Prod., cold-rolled, semi-finished Molyblenum (Mo II, Mo IV, Mo V, Mo VI) secondary 1.39e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Cadmium cmpds secondary 1.39e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Cadmium cmpds secondary	PMMA Sheet Prod.	Aluminum (+3)	secondary	3.33e-06	6.42e-05
Styrene-butadiene Copolymer Prod. Nitrate secondary 2.53e-06 4.88e-05 Aluminum Prod. Benzene secondary 2.23e-06 4.88e-05 Natural Gas Prod. Nitrates/nitrites secondary 2.10e-06 4.05e-05 Steel Prod., cold-rolled, seni-finished Phenol secondary 1.74e-06 3.35e-05 Steel Prod., cold-rolled, seni-finished Benzene secondary 1.74e-06 3.35e-05 Aluminum Prod. Toluene secondary 1.74e-06 3.35e-05 Steel Prod., cold-rolled, seni-finished Chorine secondary 1.74e-06 2.84e-05 Steel Prod., cold-rolled, seni-finished Mainum (r-3) secondary 1.74e-06 2.84e-05 Steel Prod., cold-rolled, seni-finished Molybehum (Mo II, Mo IV, Mo V, Mo V) secondary 1.41e-06 2.72e-05 Streel Prod., cold-rolled, seni-finished Molybehum (Mo II, Mo IV, Mo V, Mo V) secondary 1.38e-06 2.47e-05 Streel Prod., cold-rolled, seni-finished Sacindary 1.28e-06 2.47e-05 Steel Prod., cold-rolled, seni-finished Boriz acid secondary	Aluminum Prod.	Arsenic cmpds	secondary	3.19e-06	6.15e-05
Aluminum Prod. Benzene secondary 2.55e-06 4.88e-05 PMMA Sheet Prod. Nitrates/nitrites secondary 2.12e-06 4.28e-05 Steel Prod., cold-rolled, semi-finished Phenol secondary 1.74e-06 3.35e-05 Steel Prod., cold-rolled, semi-finished Benzene secondary 1.58e-06 3.04e-05 Steel Prod., cold-rolled, semi-finished Chlorine secondary 1.58e-06 3.04e-05 Steel Prod., cold-rolled, semi-finished Chlorine secondary 1.47e-06 2.72e-05 Steel Prod., cold-rolled, semi-finished Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo V) secondary 1.31e-06 2.57e-05 Steel Prod., cold-rolled, semi-finished Cadmium empds secondary 1.18e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Cadmium cmpds secondary 1.18e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Cadmium cmpds secondary 1.18e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Cyanide (-1) secondary 1.18e-06 2.27e-05 Steel Prod., cold-rolled, semi-finis	Styrene-butadiene Copolymer Prod.	Nitrate	secondary	2.53e-06	4.88e-05
PMMA Sheet Prod. Nitrates/initries secondary 2.22c-06 4.28e-05 Natural Gas Prod. Nitrate secondary 2.10e-06 4.05e-05 Steel Prod., cold-rolled, semi-finished Benzene secondary 1.74e-06 3.35e-05 Aluminum Prod. Toluene secondary 1.58e-06 3.04e-05 Steel Prod., cold-rolled, semi-finished Chlorine secondary 1.47e-06 2.84e-05 Aluminum Prod. Phenol secondary 1.47e-06 2.84e-05 Steel Prod., cold-rolled, semi-finished Chlorine secondary 1.41e-06 2.72e-05 Steel Prod., cold-rolled, semi-finished Cadmium empds secondary 1.31e-06 2.53e-05 Steel Prod., cold-rolled, semi-finished Cadmium empds secondary 1.31e-06 2.24e-05 Steel Prod., cold-rolled, semi-finished Cyanide (-1) secondary 1.16e-06 2.24e-05 Steel Prod., cold-rolled, semi-finished Maganese empds secondary 1.08e-06 2.16e-05 Steel Prod., cold-rolled, semi-finished Maganese empds secondary	Aluminum Prod.	Benzene	secondary	2.53e-06	4.88e-05
Natural Gas Prod. Nitrate secondary 2.10e-06 4.05e-05 Steel Prod., cold-rolled, semi-finished Phenol secondary 1.74e-06 3.35e-05 Steel Prod., cold-rolled, semi-finished Benzene secondary 1.58e-06 3.04e-05 Aluminum Prod. Aluminum (+3) secondary 1.58e-06 3.04e-05 Steel Prod., cold-rolled, semi-finished Chlorine secondary 1.41e-06 2.72e-05 Steel Prod., cold-rolled, semi-finished Chlorine secondary 1.39e-06 2.69e-05 Steel Prod., cold-rolled, semi-finished Cadmium cmpds secondary 1.38e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Cadmium cmpds secondary 1.38e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Coratic acid secondary 1.18e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Coratic acid secondary 1.08e-06 2.16e-05 Steel Prod., cold-rolled, semi-finished Coratic acid secondary 1.08e-06 2.16e-05 Steel Prod., cold-rolled, semi-finished <td< td=""><td>PMMA Sheet Prod.</td><td>Nitrates/nitrites</td><td>secondary</td><td>2.22e-06</td><td>4.28e-05</td></td<>	PMMA Sheet Prod.	Nitrates/nitrites	secondary	2.22e-06	4.28e-05
Sted Prod., cold-rolled, semi-finished Phenol secondary 1.74c-06 3.35c-05 Sted Prod., cold-rolled, semi-finished Benzene secondary 1.74c-06 3.35c-05 Aluminum Prod. Tolucne secondary 1.58c-06 3.04c-05 Sted Prod., cold-rolled, semi-finished Chlorine secondary 1.47c-06 2.84e-05 Sted Prod., cold-rolled, semi-finished Chlorine secondary 1.41e-06 2.72e-05 Steel Prod., cold-rolled, semi-finished Cadmium cmpds secondary 1.31e-06 2.53e-05 Steel Prod., cold-rolled, semi-finished Cadmium cmpds secondary 1.31e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Cyanide (-1) secondary 1.16e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Boric acid secondary 1.06e-06 2.16e-05 Steel Prod., cold-rolled, semi-finished Maganese cmpds secondary 1.06e-06 2.16e-05 Steel Prod., cold-rolled, semi-finished Nickel cmpds secondary 1.06e-05 2.10e-05 Steel Prod., cold-rolled, semi-finished	Natural Gas Prod.	Nitrate	secondary	2.10e-06	4.05e-05
Sted Prod., cold-rolled, semi-finishedBenzenesecondary1.74e-063.35e-05Aluminum Prod.Toluenesecondary1.58e-063.04e-05PET Resin ProductionAluminum (+3)secondary1.47e-062.84e-05Sted Prod., cold-rolled, semi-finishedChlorinesecondary1.41e-062.27e-05Steel Prod., cold-rolled, semi-finishedMolybdenum (Mo II, Mo II, Mo IV, Mo V) secondary1.39e-062.69e-05Steel Prod., cold-rolled, semi-finishedCadmium cmpdssecondary1.31e-062.25e-05Steel Prod., cold-rolled, semi-finishedCadmium cmpdssecondary1.18e-062.27e-05Steel Prod., cold-rolled, semi-finishedCamiade (-1)secondary1.18e-062.27e-05Steel Prod., cold-rolled, semi-finishedBoric acidsecondary1.11e-062.15e-05Steel Prod., cold-rolled, semi-finishedManganese cmpdssecondary1.08e-062.10e-05Steel Prod., cold-rolled, semi-finishedManganese cmpdssecondary1.08e-061.08e-05Steel Prod., cold-rolled, semi-finishedKickl empdssecondary1.08e-071.66e-05Steel Prod., cold-rolled, semi-finishedKickl empdssecondary8.61e-071.66e-05Steel Prod., cold-rolled, semi-finishedNickle (rupdssecondary5.8ee-079.95e-06Aluminum Prod.Nickle (+2)secondary5.16e-079.95e-06Aluminum Prod.Nickle (+2)secondary2.91e-075.61e-07Polycarbonate ProductionNickl	Steel Prod., cold-rolled, semi-finished	Phenol	secondary	1.74e-06	3.35e-05
Aluminum Prod. Toluene secondary 1.58e-06 3.04e-05 PET Resin Production Aluminum (+3) secondary 1.47e-06 2.84e-05 Sted Prod., cold-rolled, semi-finished Chlorine secondary 1.47e-06 2.84e-05 Sted Prod., cold-rolled, semi-finished Molybdenum (Mo II, Mo III, Mo IV, Mo V) secondary 1.39e-06 2.69e-05 Sted Prod., cold-rolled, semi-finished Molybdenum (mo II, Mo III, Mo IV, Mo V) secondary 1.39e-06 2.47e-05 Sted Prod., cold-rolled, semi-finished Cadmium cmpds secondary 1.8e-06 2.27e-05 Sted Prod., cold-rolled, semi-finished Cyanide (-1) secondary 1.16e-06 2.27e-05 Sted Prod., cold-rolled, semi-finished Toluene secondary 1.16e-06 2.12e-05 Sted Prod., cold-rolled, semi-finished Toluene secondary 1.09e-06 2.10e-05 Sted Prod., cold-rolled, semi-finished Maganese cmpds secondary 9.15e-07 1.76e-05 Sted Prod., cold-rolled, semi-finished Chromium (V1) secondary 8.61e-07 9.05e-06 Sted Prod., cold-rolled, semi-finish	Steel Prod., cold-rolled, semi-finished	Benzene	secondary	1.74e-06	3.35e-05
PET Resin Production Aluminum (+3) secondary 1.58c-06 3.04c-05 Steel Prod., cold-rolled, semi-finished Chlorine secondary 1.47c-06 2.242c-05 Aluminum Prod. Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI) secondary 1.39c-06 2.72c-05 Steel Prod., cold-rolled, semi-finished Cadmium empds secondary 1.31c-06 2.53c-05 Steel Prod., cold-rolled, semi-finished Cadmium empds secondary 1.28c-06 2.247c-05 Steel Prod., cold-rolled, semi-finished Cyanide (-1) secondary 1.18c-06 2.24rc-05 Steel Prod., cold-rolled, semi-finished Boric acid secondary 1.10c-06 2.10c-05 Steel Prod., cold-rolled, semi-finished Toluene secondary 1.08c-06 2.10c-05 Steel Prod., cold-rolled, semi-finished Manganese empds secondary 1.08c-06 2.10c-05 Steel Prod., cold-rolled, semi-finished Ethylbenzene secondary 1.66-07 1.06c-05 Steel Prod., cold-rolled, semi-finished Nickel empds secondary 5.8c-07 1.13c-05 Steel Prod., cold-r	Aluminum Prod.	Toluene	secondary	1.58e-06	3.04e-05
Steel Prod., cold-rolled, semi-finished Chlorine secondary 1.47e-06 2.84e-05 Aluminum Prod. Phenol secondary 1.41e-06 2.72e-05 Steel Prod., cold-rolled, semi-finished Molybdenum (Mo II, Mo IV, Mo V, Mo VI) secondary 1.39e-06 2.63e-05 Steel Prod., cold-rolled, semi-finished Cadmium empds secondary 1.28e-06 2.53e-05 Styrene-butadiene Copolymer Prod. Phenol secondary 1.18e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Coyanide (-1) secondary 1.18e-06 2.24e-05 Steel Prod., cold-rolled, semi-finished Boric acid secondary 1.09e-06 2.10e-05 Steel Prod., cold-rolled, semi-finished Manganese cmpds secondary 1.03e-06 1.08e-05 Steel Prod., cold-rolled, semi-finished Manganese cmpds secondary 1.03e-06 1.08e-05 Steel Prod., cold-rolled, semi-finished Chromium (V1) secondary 5.06e-07 1.16e-05 Steel Prod., cold-rolled, semi-finished Chromium (V1) secondary 5.06e-07 9.95e-06 Aluminum Prod. <td>PET Resin Production</td> <td>Aluminum (+3)</td> <td>secondary</td> <td>1.58e-06</td> <td>3.04e-05</td>	PET Resin Production	Aluminum (+3)	secondary	1.58e-06	3.04e-05
Aluminum Prod. Phenol secondary 1.41e-06 2.72e-05 Steel Prod., cold-rolled, semi-finished Molybdenum (Mo II, Mo IV, Mo V), Secondary 1.39e-06 2.69e-05 Steel Prod., cold-rolled, semi-finished Cadmium empds secondary 1.8e-06 2.27e-05 Streen Prod. Phenol secondary 1.8e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Cyanide (-1) secondary 1.16e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Boric acid secondary 1.10e-06 2.27e-05 Steel Prod., cold-rolled, semi-finished Boric acid secondary 1.08e-06 2.16e-05 Steel Prod., cold-rolled, semi-finished Manganese empds secondary 1.03e-06 1.08e-05 Steel Prod., cold-rolled, semi-finished Ethylbenzene secondary 8.61e-07 1.16e-05 Steel Prod., cold-rolled, semi-finished Chromium (V1) secondary 5.16e-07 9.95e-06 Polycarbonate Production Nickel empds secondary 5.16e-07 9.95e-06 Polycarbonate Production Nickel (+2)	Steel Prod., cold-rolled, semi-finished	Chlorine	secondary	1.47e-06	2.84e-05
Steel Prod., cold-rolled, semi-finishedMolybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI) secondary1.39e-062.69e-05Steel Prod., cold-rolled, semi-finishedCadmum cmpdssecondary1.31e-062.53e-05Styene-butadiene Copolymer Prod.Phenolsecondary1.28e-062.47e-05PET Resin ProductionPhenolsecondary1.16e-062.27e-05Steel Prod., cold-rolled, semi-finishedCyanide (-1)secondary1.11e-062.27e-05Steel Prod., cold-rolled, semi-finishedBoric acidsecondary1.09e-062.10e-05Steel Prod., cold-rolled, semi-finishedManganese cmpdssecondary1.09e-061.98e-05PMMA Sheet Prod.Nickel empdssecondary9.15e-071.76e-05Steel Prod., cold-rolled, semi-finishedChromium (V1)secondary8.61e-071.16e-05Steel Prod., cold-rolled, semi-finishedNickel empdssecondary5.85e-071.13e-05Polycarbonate ProductionNickel (+2)secondary5.16e-079.95e-06Polycarbonate ProductionNitratesecondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedChromium (III)secondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedTitanium tetrachloridesecondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedChromium (III)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTitanium tetrachloridesecondary2.91e-075.61e-06 <tr< td=""><td>Aluminum Prod.</td><td>Phenol</td><td>secondary</td><td>1.41e-06</td><td>2.72e-05</td></tr<>	Aluminum Prod.	Phenol	secondary	1.41e-06	2.72e-05
Steel Prod., cold-rolled, semi-finishedCadmium empdssecondary1.31e-062.53e-05Styrene-butadiene Copolymer Prod.Phenolsecondary1.28e-062.47e-05PET Resin ProductionPhenolsecondary1.18e-062.27e-05Steel Prod., cold-rolled, semi-finishedBoric acidsecondary1.11e-062.24e-05Steel Prod., cold-rolled, semi-finishedBoric acidsecondary1.09e-062.10e-05Steel Prod., cold-rolled, semi-finishedManganese empdssecondary1.03e-061.98e-05PMM Sheet Prod.Nickel empdssecondary9.15e-071.76e-05Steel Prod., cold-rolled, semi-finishedChromium (VI)secondary6.04e-071.16e-05Steel Prod., cold-rolled, semi-finishedNickel empdssecondary5.85e-071.13e-05Polycarbonate ProductionNickel empdssecondary5.16e-079.95e-06Polycarbonate ProductionNitratesecondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedVanatium (V3+, V5+)secondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedTitanium (V1)secondary2.91e-075.61e-07Steel Prod., cold-rolled, semi-finishedVanatium (V3+, V5+)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTitanium (V1)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTitanium (V1)secondary3.93e-077.58e-06Steel Prod., cold-rolled, se	Steel Prod., cold-rolled, semi-finished	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI)	secondary	1.39e-06	2.69e-05
Styrene-butadiene Copolymer Prod.Phenolsecondary1.28e-062.27e-05PET Resin ProductionPhenolsecondary1.18e-062.27e-05Steel Prod., cold-rolled, semi-finishedBoric acidsecondary1.11e-062.24e-05Steel Prod., cold-rolled, semi-finishedBoric acidsecondary1.11e-062.15e-05Steel Prod., cold-rolled, semi-finishedManganese empdssecondary1.09e-062.10e-05Steel Prod., cold-rolled, semi-finishedManganese empdssecondary9.15e-071.76e-05Steel Prod., cold-rolled, semi-finishedEthylbenzenesecondary8.61e-071.66e-05Steel Prod., cold-rolled, semi-finishedKickl empdssecondary5.86e-071.16e-05Steel Prod., cold-rolled, semi-finishedNickle mpdssecondary5.86e-071.95e-06Steel Prod., cold-rolled, semi-finishedNickle (rupdssecondary5.16e-079.95e-06Polycarbonate ProductionNitritessecondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedChromium (VI)secondary3.93e-077.58e-06Aluminum Prod.Nitritessecondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTanium (VI)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTianium (VI)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTianium (VI)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-fin	Steel Prod., cold-rolled, semi-finished	Cadmium cmpds	secondary	1.31e-06	2.53e-05
PET Resin ProductionPhenolsecondary1.18e-062.27e-05Steel Prod., cold-rolled, semi-finishedCyanide (-1)secondary1.16e-062.24e-05Steel Prod., cold-rolled, semi-finishedBoric acidsecondary1.11e-062.15e-05Steel Prod., cold-rolled, semi-finishedToluenesecondary1.09e-062.10e-05Steel Prod., cold-rolled, semi-finishedMaganese empdssecondary9.15e-071.76e-05PMAA Sheet Prod.Nickel empdssecondary8.61e-071.66e-05Steel Prod., cold-rolled, semi-finishedEthylbenzenesecondary6.04e-071.16e-05Steel Prod., cold-rolled, semi-finishedChromium (VI)secondary5.85e-071.13e-05Steel Prod., cold-rolled, semi-finishedNickel empdssecondary5.16e-079.95e-06Polycarbonate ProductionNickel (+2)secondary5.16e-079.95e-06Polycarbonate ProductionNitratesecondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedChromium (III)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTitanium tetrachloridesecondary2.92e-074.99e-06Aluminum Prod.Ohromium (VI)secondary2.14e-074.13e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-	Styrene-butadiene Copolymer Prod.	Phenol	secondary	1.28e-06	2.47e-05
Steel Prod., cold-rolled, semi-finishedCyanide (-1)secondary1.16e-062.24e-05Steel Prod., cold-rolled, semi-finishedBoric acidsecondary1.09e-062.10e-05Steel Prod., cold-rolled, semi-finishedManganese cmpdssecondary1.03e-061.98e-05PMMA Sheet Prod.Nickel empdssecondary9.15e-071.76e-05Steel Prod., cold-rolled, semi-finishedEthylbenzenesecondary8.61e-071.66e-05Steel Prod., cold-rolled, semi-finishedChromium (VI)secondary6.04e-071.16e-05Steel Prod., cold-rolled, semi-finishedNickel empdssecondary5.85e-071.13e-05Polycarbonate ProductionNickel empdssecondary5.16e-079.95e-06Polycarbonate ProductionNitratesecondary5.16e-079.95e-06Aluminum Prod.Nitritessecondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedChromium (III)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedChromium (III)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedChromium (VI)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedChromium (VI)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedChromium (VI)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedChromium (VI)secondary3.93e-073.50e-06Steel Prod., cold-rolled, s	PET Resin Production	Phenol	secondary	1.18e-06	2.27e-05
Steel Prod., cold-rolled, semi-finishedBoric acidsecondary1.11e-062.15e-05Steel Prod., cold-rolled, semi-finishedToluenesecondary1.09e-062.10e-05Steel Prod., cold-rolled, semi-finishedManganese cmpdssecondary1.03e-061.98e-05PMMA Sheet Prod.Nickel cmpdssecondary9.15e-071.76e-05Steel Prod., cold-rolled, semi-finishedEthylbenzenesecondary8.61e-071.66e-05Steel Prod., cold-rolled, semi-finishedChromium (VI)secondary6.04e-071.16e-05Steel Prod., cold-rolled, semi-finishedNickel cmpdssecondary5.85e-071.13e-05Polycarbonate ProductionNickel (+2)secondary5.16e-079.95e-06Polycarbonate ProductionNitratesecondary5.16e-079.95e-06Steel Prod., cold-rolled, semi-finishedChromium (III)secondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedVanadium (V3+, V5+)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTitanium tetrachloridesecondary2.59e-074.99e-06Attural Gas Prod.Chromium (VI)secondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-0	Steel Prod., cold-rolled, semi-finished	Cyanide (-1)	secondary	1.16e-06	2.24e-05
Steel Prod., cold-rolled, semi-finishedToluenesecondary1.09e-062.10e-05Steel Prod., cold-rolled, semi-finishedManganese cmpdssecondary1.03e-061.98e-05PMMA Sheet Prod.Nickel cmpdssecondary9.15e-071.76e-05Steel Prod., cold-rolled, semi-finishedEthylbenzenesecondary6.04e-071.16e-05Steel Prod., cold-rolled, semi-finishedChromium (VI)secondary6.04e-071.16e-05Steel Prod., cold-rolled, semi-finishedNickel cmpdssecondary5.85e-071.13e-05Polycarbonate ProductionNickel (+2)secondary5.16e-079.95e-06Polycarbonate ProductionNitratesecondary3.93e-077.58e-06Aluminum Prod.Nitritessecondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedChromium (III)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedVanadium (V3+, V5+)secondary2.99e-074.99e-06Atural Gas Prod.Chromium (VI)secondary2.59e-074.99e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06 <tr <tr="">PET Resin Prod</tr>	Steel Prod., cold-rolled, semi-finished	Boric acid	secondary	1.11e-06	2.15e-05
Steel Prod., cold-rolled, semi-finishedManganese cmpdssecondary1.03e-061.98e-05PMMA Sheet Prod.Nickel cmpdssecondary9.15e-071.76e-05Steel Prod., cold-rolled, semi-finishedEthylbenzenesecondary8.61e-071.16e-05Steel Prod., cold-rolled, semi-finishedChromium (VI)secondary6.04e-071.16e-05Steel Prod., cold-rolled, semi-finishedNickel cmpdssecondary5.85e-071.13e-05Polycarbonate ProductionNickel (+2)secondary5.16e-079.95e-06Polycarbonate ProductionNitratesecondary3.16e-079.95e-06Aluminum Prod.Nitritessecondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedChromium (III)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTitanium tetrachloridesecondary2.59e-074.99e-06Atural Gas Prod.Chromium (VI)secondary2.14e-074.13e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNicke	Steel Prod., cold-rolled, semi-finished	Toluene	secondary	1.09e-06	2.10e-05
PMMA Sheet Prod.Nickel cmpdssecondary9.15e-071.76e-05Steel Prod., cold-rolled, semi-finishedEthylbenzenesecondary8.61e-071.66e-05Steel Prod., cold-rolled, semi-finishedChromium (VI)secondary6.04e-071.16e-05Steel Prod., cold-rolled, semi-finishedNickel cmpdssecondary5.85e-071.13e-05Polycarbonate ProductionNickel (+2)secondary5.16e-079.95e-06Aluminum Prod.Nitritessecondary4.67e-079.01e-06Steel Prod., cold-rolled, semi-finishedChromium (III)secondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedVanadium (V3+, V5+)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTitanium tetrachloridesecondary2.59e-074.99e-06Natural Gas Prod.Chromium (VI)secondary2.59e-074.99e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.81e-073.49e-06Natural Gas Prod.Chromium (III)secondary1.81e-073.49e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.81e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.81e-073.50e-06 <td>Steel Prod., cold-rolled, semi-finished</td> <td>Manganese cmpds</td> <td>secondary</td> <td>1.03e-06</td> <td>1.98e-05</td>	Steel Prod., cold-rolled, semi-finished	Manganese cmpds	secondary	1.03e-06	1.98e-05
Steel Prod., cold-rolled, semi-finishedEthylbenzenesecondary8.61e-071.66e-05Steel Prod., cold-rolled, semi-finishedChromium (VI)secondary6.04e-071.16e-05Steel Prod., cold-rolled, semi-finishedNickel cmpdssecondary5.85e-071.13e-05Polycarbonate ProductionNickel (+2)secondary5.16e-079.95e-06Aluminum Prod.Nitratesecondary5.16e-079.01e-06Steel Prod., cold-rolled, semi-finishedChromium (III)secondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedVanadium (V3+, V5+)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTitanium tetrachloridesecondary2.59e-074.99e-06Natural Gas Prod.Chromium (VI)secondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.82e-073.50e-06Natural Gas Prod.Chromium (III)secondary1.84e-072.72e-06Natural Gas Prod.Chromium (III)secondary1.84e-072.72e-06Natural Gas Prod.Chromium (III)secondary1.54e-072.96e-06Natural Gas Prod.Chromium (III)secondary1.54e-072.96e-06Natural Gas Pr	PMMA Sheet Prod.	Nickel cmpds	secondary	9.15e-07	1.76e-05
Steel Prod., cold-rolled, semi-finishedChromium (VI)secondary6.04e-071.16e-05Steel Prod., cold-rolled, semi-finishedNickel cmpdssecondary5.85e-071.13e-05Polycarbonate ProductionNickel (+2)secondary5.16e-079.95e-06Aluminum Prod.Nitratesecondary5.16e-079.95e-06Aluminum Prod.Nitritessecondary4.67e-079.01e-06Steel Prod., cold-rolled, semi-finishedChromium (III)secondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedVanadium (V3+, V5+)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTitanium tetrachloridesecondary2.59e-074.99e-06Natural Gas Prod.Chromium (VI)secondary1.82e-073.50e-06PET Resin ProductionMickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNitratesecondary1.82e-073.50e-06PET Resin ProductionNitratesecondary1.82e-073.50e-06PET Resin ProductionNitratesecondary1.82e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.82e-073.50e-06Natural Gas Prod.Chromium (III)secondary1.81e-072.96e-06Natural Gas Prod.Chromium (III)secondary1.54e-072.96e-06Natural Gas Prod.Chromium (III)secondary1.54e-072.96e-06Natural Gas Prod.Chromium (III)second	Steel Prod., cold-rolled, semi-finished	Ethylbenzene	secondary	8.61e-07	1.66e-05
Steel Prod., cold-rolled, semi-finishedNickel cmpdssecondary5.85e-071.13e-05Polycarbonate ProductionNickel (+2)secondary5.16e-079.95e-06Polycarbonate ProductionNitratesecondary5.16e-079.95e-06Aluminum Prod.Nitritessecondary4.67e-079.01e-06Steel Prod., cold-rolled, semi-finishedChromium (III)secondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedVanadium (V3+, V5+)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTitanium tetrachloridesecondary2.59e-074.99e-06Natural Gas Prod.Chromium (VI)secondary2.14e-074.13e-06PET Resin ProductionMicrurysecondary1.82e-073.50e-06PET Resin ProductionNitratesecondary1.82e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.82e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.82e-073.50e-06Natural Gas Prod.Chromium (III)secondary1.81e-073.49e-06Natural Gas Prod.Chromium (III)secondary1.54e-072.96e-06Aluminum Prod.Cyanide (-1)secondary1.54e-072.96e-06Aluminum Prod.Cyanide (-1)secondary1.41e-072.72e-06Steel Prod. cold-rolled semi-finishedArsenic cmpdssecondary1.36e-072.62e-06	Steel Prod., cold-rolled, semi-finished	Chromium (VI)	secondary	6.04e-07	1.16e-05
Polycarbonate ProductionNickel (+2)secondary5.16e-079.95e-06Polycarbonate ProductionNitratesecondary5.16e-079.95e-06Aluminum Prod.Nitritessecondary4.67e-079.01e-06Steel Prod., cold-rolled, semi-finishedChromium (III)secondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedVanadium (V3+, V5+)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTitanium tetrachloridesecondary2.59e-074.99e-06Natural Gas Prod.Chromium (VI)secondary2.14e-074.13e-06PET Resin ProductionMercurysecondary1.82e-073.50e-06PET Resin ProductionNitratesecondary1.82e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.82e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.81e-073.49e-06Natural Gas Prod.Chromium (III)secondary1.54e-072.96e-06Aluminum Prod.Chyaide (-1)secondary1.54e-072.96e-06Steel Prod. cold-rolled semi-finishedArsenic cmndssecondary1.36e-072.72e-06	Steel Prod., cold-rolled, semi-finished	Nickel cmpds	secondary	5.85e-07	1.13e-05
Polycarbonate ProductionNitratesecondary5.16e-079.95e-06Aluminum Prod.Nitritessecondary4.67e-079.01e-06Steel Prod., cold-rolled, semi-finishedChromium (III)secondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedVanadium (V3+, V5+)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTitanium tetrachloridesecondary2.59e-074.99e-06Natural Gas Prod.Chromium (VI)secondary2.14e-074.13e-06PET Resin ProductionMercurysecondary1.82e-073.50e-06PET Resin ProductionNitratesecondary1.82e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.81e-073.49e-06Natural Gas Prod.Chromium (III)secondary1.54e-072.96e-06Styrene-butadiene Copolymer Prod.Nickel (-1)secondary1.54e-072.96e-06Aluminum Prod.Cyanide (-1)secondary1.41e-072.72e-06Steel Prod. cold-rolled semi-finishedArsenic cmndssecondary1.36e-072.62e-06	Polycarbonate Production	Nickel (+2)	secondary	5.16e-07	9.95e-06
Aluminum Prod.Nitritessecondary4.67e-079.01e-06Steel Prod., cold-rolled, semi-finishedChromium (III)secondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedVanadium (V3+, V5+)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTitanium tetrachloridesecondary2.59e-074.99e-06Natural Gas Prod.Chromium (VI)secondary2.14e-074.13e-06PET Resin ProductionMercurysecondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.82e-073.50e-06Natural Gas Prod.Chromium (III)secondary1.81e-073.49e-06Natural Gas Prod.Chromium (III)secondary1.54e-072.96e-06Aluminum Prod.Cyanide (-1)secondary1.41e-072.72e-06Steel Prod. cold-rolled semi-finishedArsenic cmndssecondary1.36e-072.62e-06	Polycarbonate Production	Nitrate	secondary	5.16e-07	9.95e-06
Steel Prod., cold-rolled, semi-finishedChromium (III)secondary3.93e-077.58e-06Steel Prod., cold-rolled, semi-finishedVanadium (V3+, V5+)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTitanium tetrachloridesecondary2.59e-074.99e-06Natural Gas Prod.Chromium (VI)secondary2.14e-074.13e-06PET Resin ProductionMercurysecondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.82e-073.50e-06Natural Gas Prod.Chromium (III)secondary1.81e-073.49e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.81e-072.96e-06Aluminum Prod.Cyanide (-1)secondary1.41e-072.72e-06Steel Prod. cold-rolled semi-finishedArsenic cmndssecondary1.36e-072.62e-06	Aluminum Prod.	Nitrites	secondary	4.67e-07	9.01e-06
Steel Prod., cold-rolled, semi-finishedVanadium (V3+, V5+)secondary2.91e-075.61e-06Steel Prod., cold-rolled, semi-finishedTitanium tetrachloridesecondary2.59e-074.99e-06Natural Gas Prod.Chromium (VI)secondary2.14e-074.13e-06PET Resin ProductionMercurysecondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNitratesecondary1.82e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.81e-073.49e-06Natural Gas Prod.Chromium (III)secondary1.54e-072.96e-06Aluminum Prod.Cyanide (-1)secondary1.41e-072.72e-06Steel Prod. cold-rolled semi-finishedArsenic cmndssecondary1.36e-072.62e-06	Steel Prod., cold-rolled, semi-finished	Chromium (III)	secondary	3.93e-07	7.58e-06
Steel Prod., cold-rolled, semi-finishedTitanium tetrachloridesecondary2.59e-074.99e-06Natural Gas Prod.Chromium (VI)secondary2.14e-074.13e-06PET Resin ProductionMercurysecondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNitratesecondary1.82e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.82e-073.50e-06Natural Gas Prod.Chromium (III)secondary1.81e-073.49e-06Aluminum Prod.Cyanide (-1)secondary1.41e-072.72e-06Steel Prod. cold-rolled semi-finishedArsenic cmndssecondary1.36e-072.62e-06	Steel Prod., cold-rolled, semi-finished	Vanadium (V3+, V5+)	secondary	2.91e-07	5.61e-06
Natural Gas Prod.Chromium (VI)secondary2.14e-074.13e-06PET Resin ProductionMercurysecondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNitratesecondary1.82e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.81e-073.49e-06Natural Gas Prod.Chromium (III)secondary1.54e-072.96e-06Aluminum Prod.Cyanide (-1)secondary1.41e-072.72e-06Steel Prod. cold-rolled semi-finishedArsenic cmndssecondary1.36e-072.62e-06	Steel Prod., cold-rolled, semi-finished	Titanium tetrachloride	secondary	2.59e-07	4.99e-06
PET Resin ProductionMercurysecondary1.82e-073.50e-06PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNitratesecondary1.82e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.81e-073.49e-06Natural Gas Prod.Chromium (III)secondary1.54e-072.96e-06Aluminum Prod.Cyanide (-1)secondary1.41e-072.72e-06Steel Prod. cold-rolled semi-finishedArsenic cmrdssecondary1.36e-072.62e-06	Natural Gas Prod.	Chromium (VI)	secondary	2.14e-07	4.13e-06
PET Resin ProductionNickel (+2)secondary1.82e-073.50e-06PET Resin ProductionNitratesecondary1.82e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.81e-073.49e-06Natural Gas Prod.Chromium (III)secondary1.54e-072.96e-06Aluminum Prod.Cyanide (-1)secondary1.41e-072.72e-06Steel Prod. cold-rolled semi-finishedArsenic cmrdssecondary1.36e-072.62e-06	PET Resin Production	Mercury	secondary	1.82e-07	3.50e-06
PET Resin ProductionNitratesecondary1.82e-073.50e-06Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.81e-073.49e-06Natural Gas Prod.Chromium (III)secondary1.54e-072.96e-06Aluminum Prod.Cyanide (-1)secondary1.41e-072.72e-06Steel Prod. cold-rolled semi-finishedArsenic cmndssecondary1.36e-072.62e-06	PET Resin Production	Nickel (+2)	secondary	1.82e-07	3.50e-06
Styrene-butadiene Copolymer Prod.Nickel (+2)secondary1.81e-073.49e-06Natural Gas Prod.Chromium (III)secondary1.54e-072.96e-06Aluminum Prod.Cyanide (-1)secondary1.41e-072.72e-06Steel Prod. cold-rolled semi-finishedArsenic cmndssecondary1.36e-072.62e-06	PET Resin Production	Nitrate	secondary	1.82e-07	3.50e-06
Natural Gas Prod.Chromium (III)secondary1.54e-072.96e-06Aluminum Prod.Cyanide (-1)secondary1.41e-072.72e-06Steel Prod. cold-rolled semi-finishedArsenic cmndssecondary1.36e-072.62e-06	Styrene-butadiene Copolymer Prod.	Nickel (+2)	secondary	1.81e-07	3.49e-06
Aluminum Prod.Cyanide (-1)secondary1.41e-072.72e-06Steel Prod. cold-rolled semi-finishedArsenic cmndssecondary1.36e-072.62e-06	Natural Gas Prod.	Chromium (III)	secondary	1.54e-07	2.96e-06
Steel Prod_cold-rolled_semi-finished_Arsenic cmndssecondary136e_07262e_06	Aluminum Prod.	Cvanide (-1)	secondary	1.41e-07	2.72e-06
	Steel Prod., cold-rolled, semi-finished	Arsenic cmpds	secondary	1.36e-07	2.62e-06

			Aquatic Toxicity	
Process Group	Material	LCI Data Type	(tox-kg)	% of Total
Steel Prod., cold-rolled, semi-finished	Rubidium ion (Rb+)	secondary	1.19e-07	2.29e-06
Steel Prod., cold-rolled, semi-finished	Nitrates/nitrites	secondary	9.51e-08	1.83e-06
Steel Prod., cold-rolled, semi-finished	Selenium	secondary	7.81e-08	1.51e-06
Steel Prod., cold-rolled, semi-finished	Boron (B III)	secondary	3.45e-08	6.65e-07
Natural Gas Prod.	Copper (+1 & +2)	secondary	2.19e-08	4.22e-07
Steel Prod., cold-rolled, semi-finished	Phosphorus pentoxide	secondary	1.53e-08	2.94e-07
Steel Prod., cold-rolled, semi-finished	Hydrazine	secondary	1.13e-08	2.18e-07
Steel Prod., cold-rolled, semi-finished	Morpholine	secondary	9.20e-09	1.77e-07
Aluminum Prod.	Chromium (VI)	secondary	3.43e-09	6.61e-08
Steel Prod., cold-rolled, semi-finished	Hypochlorous acid	secondary	1.72e-09	3.33e-08
Steel Prod., cold-rolled, semi-finished	Cobalt (Co I, Co II, Co III)	secondary	1.54e-09	2.97e-08
Steel Prod., cold-rolled, semi-finished	Dichloromethane	secondary	9.72e-10	1.87e-08
Steel Prod., cold-rolled, semi-finished	Triethylene glycol	secondary	6.72e-10	1.29e-08
Aluminum Prod.	Triethylene glycol	secondary	6.46e-10	1.24e-08
Natural Gas Prod.	Aromatic hydrocarbons	secondary	2.47e-10	4.77e-09
Steel Prod., cold-rolled, semi-finished	Nitrites	secondary	2.36e-10	4.55e-09
Natural Gas Prod.	Barium cmpds	secondary	2.01e-10	3.87e-09
Natural Gas Prod.	Cadmium cmpds	secondary	9.72e-11	1.87e-09
Steel Prod., cold-rolled, semi-finished	Edetic acid (EDTA)	secondary	6.24e-11	1.20e-09
Natural Gas Prod.	Lead cmpds	secondary	3.25e-11	6.26e-10
Natural Gas Prod.	Toluene	secondary	3.02e-11	5.82e-10
Natural Gas Prod.	Nickel cmpds	secondary	3.93e-12	7.58e-11
Natural Gas Prod.	Mercury compounds	secondary	1.77e-12	3.40e-11
Steel Prod., cold-rolled, semi-finished	Chloroform	secondary	1.57e-12	3.03e-11
Steel Prod., cold-rolled, semi-finished	Lithium salts	secondary	1.17e-12	2.25e-11
Natural Gas Prod.	Cyanide (-1)	secondary	1.15e-12	2.22e-11
Steel Prod., cold-rolled, semi-finished	Trichloroethylene	secondary	7.83e-13	1.51e-11
Natural Gas Prod.	Halogenated matter (organic)	secondary	3.29e-13	6.35e-12
Steel Prod., cold-rolled, semi-finished	Formaldehyde	secondary	5.77e-14	1.11e-12
Steel Prod., cold-rolled, semi-finished	Tetrachloroethylene	secondary	4.21e-14	8.13e-13
Steel Prod., cold-rolled, semi-finished	1,1,1-Trichloroethane	secondary	2.91e-14	5.62e-13
Steel Prod., cold-rolled, semi-finished	Hexachloroethane	secondary	1.75e-16	3.38e-15
Total Materials Processing			6.19e-02	1.19e+00
Manufacturing Life-cycle Stage				
Monitor/module	Phosphorus (yellow or white)	primary	5.06e+00	9.77e+01
Panel components	Phosphorus (yellow or white)	primary	2.92e-02	5.63e-01
Monitor/module	Fluorides (F-)	primary	2.56e-02	4.93e-01
Monitor/module	Copper	primary	2.44e-03	4.70e-02
Monitor/module	Hexane	primary	1.18e-03	2.27e-02
Monitor/module	Zinc (elemental)	primary	9.73e-04	1.88e-02
LCD glass mfg.	Fluorides (F-)	primary	2.72e-04	5.24e-03
LPG Production	Phenol	secondary	2.04e-04	3.93e-03
LPG Production	Aluminum (+3)	secondary	1.18e-04	2.27e-03
Monitor/module	Lead	primary	1.23e-05	2.38e-04
Monitor/module	Chromium	primary	1.06e-05	2.05e-04
Monitor/module	Cyanide (-1)	primary	7.33e-06	1.41e-04
LPG Production	Nitrate	secondary	5.81e-06	1.12e-04
LCD glass mfg.	Lead	primary	4.02e-06	7.75e-05
LPG Production	Fluorides (F-)	secondary	3.27e-06	6.30e-05
		•		

			Aquatic Toxicity	
Process Group	Material	LCI Data Type	(tox-kg)	% of Total
Monitor/module	Boron	primary	1.61e-06	3.11e-05
Fuel Oil #4 Prod.	Phenol	secondary	1.49e-06	2.88e-05
Fuel Oil #4 Prod.	Aluminum (+3)	secondary	8.64e-07	1.67e-05
LPG Production	Copper (+1 & +2)	secondary	7.43e-07	1.43e-05
Monitor/module	Chromium (VI)	primary	6.39e-07	1.23e-05
Fuel Oil #2 Prod.	Phenol	secondary	5.57e-07	1.07e-05
Fuel Oil #6 Prod.	Phenol	secondary	4.90e-07	9.44e-06
Monitor/module	Antimony	primary	4.64e-07	8.95e-06
Monitor/module	Manganese	primary	4.58e-07	8.83e-06
Monitor/module	Nickel	primary	4.58e-07	8.83e-06
Monitor/module	Polychlorinated biphenyls	primary	4.07e-07	7.85e-06
LCD glass mfg.	Nitrate	primary	3.66e-07	7.06e-06
Fuel Oil #2 Prod.	Aluminum (+3)	secondary	3.22e-07	6.21e-06
Natural Gas Prod.	Zinc (+2)	secondary	3.05e-07	5.89e-06
Fuel Oil #6 Prod.	Aluminum (+3)	secondary	2.83e-07	5.46e-06
Monitor/module	Phenol	primary	2.70e-07	5.21e-06
Monitor/module	Arsenic	primary	2.28e-07	4.40e-06
Monitor/module	Cadmium	primary	2.28e-07	4.40e-06
LPG Production	Zinc (+2)	secondary	2.08e-07	4.01e-06
Monitor/module	Mercury	primary	1.94e-07	3.74e-06
Natural Gas Prod.	Fluorides (F-)	secondary	1.88e-07	3.62e-06
Natural Gas Prod.	Phenol	secondary	1.36e-07	2.63e-06
Natural Gas Prod.	Aluminum (+3)	secondary	7.83e-08	1.51e-06
Monitor/module	Tetrachloroethylene	primary	7.63e-08	1.47e-06
Natural Gas Prod.	Nitrate	secondary	4.74e-08	9.15e-07
Monitor/module	Tin	primary	4.57e-08	8.82e-07
Fuel Oil #4 Prod.	Nitrate	secondary	4.52e-08	8.72e-07
Fuel Oil #4 Prod.	Fluorides (F-)	secondary	3.51e-08	6.77e-07
Monitor/module	1,1,1-Trichloroethane	primary	2.40e-08	4.62e-07
Monitor/module	Trichloroethylene	primary	2.34e-08	4.51e-07
Fuel Oil #6 Prod.	Fluorides (F-)	secondary	1.87e-08	3.61e-07
Fuel Oil #6 Prod.	Nitrate	secondary	1.65e-08	3.19e-07
Fuel Oil #2 Prod.	Nitrate	secondary	1.61e-08	3.11e-07
Fuel Oil #2 Prod.	Fluorides (F-)	secondary	9.97e-09	1.92e-07
LPG Production	Aromatic hydrocarbons	secondary	8.40e-09	1.62e-07
LCD glass mfg.	Nickel	primary	7.60e-09	1.47e-07
LPG Production	Barium cmpds	secondary	6.82e-09	1.32e-07
Fuel Oil #4 Prod.	Copper (+1 & +2)	secondary	5.44e-09	1.05e-07
LPG Production	Chromium (VI)	secondary	4.87e-09	9.39e-08
Natural Gas Prod.	Chromium (VI)	secondary	4.83e-09	9.32e-08
LCD glass mfg.	Chromium	primary	4.57e-09	8.81e-08
LPG Production	Chromium (III)	secondary	3.49e-09	6.74e-08
Natural Gas Prod.	Chromium (III)	secondary	3.46e-09	6.68e-08
LPG Production	Cadmium cmpds	secondary	3.30e-09	6.37e-08
Fuel Oil #2 Prod.	Copper $(+1 \& +2)$	secondary	2.03e-09	3.91e-08
Fuel Oil #6 Prod.	Copper (+1 & +2)	secondary	1.78e-09	3.44e-08
Fuel Oil #4 Prod.	Zinc (+2)	secondary	1.70e-09	3.28e-08
LPG Production	Lead cmpds	secondary	1.10e-09	2.13e-08
LPG Production	Toluene	secondary	1 03e-09	1 98e-08
Fuel Oil #6 Prod.	$Z_{inc}(+2)$	secondary	6 70e-10	1.29e-08
Fuel Oil #2 Prod.	Zinc (+2)	secondary	5 84e-10	1 13e-08
	(-)		5.610 10	

			Aquatic Toxicity	
Process Group	Material	LCI Data Type	(tox-kg)	% of Total
Natural Gas Prod.	Copper (+1 & +2)	secondary	4.93e-10	9.51e-09
LPG Production	Nickel cmpds	secondary	1.34e-10	2.58e-09
Fuel Oil #4 Prod.	Aromatic hydrocarbons	secondary	6.15e-11	1.19e-09
LPG Production	Mercury compounds	secondary	6.00e-11	1.16e-09
Fuel Oil #4 Prod.	Barium cmpds	secondary	4.99e-11	9.63e-10
LPG Production	Cyanide (-1)	secondary	3.92e-11	7.56e-10
Fuel Oil #4 Prod.	Cadmium cmpds	secondary	2.42e-11	4.67e-10
Fuel Oil #2 Prod.	Aromatic hydrocarbons	secondary	2.29e-11	4.42e-10
Fuel Oil #4 Prod.	Chromium (VI)	secondary	2.06e-11	3.98e-10
Fuel Oil #6 Prod.	Aromatic hydrocarbons	secondary	2.02e-11	3.89e-10
Fuel Oil #2 Prod.	Barium cmpds	secondary	1.86e-11	3.59e-10
Fuel Oil #6 Prod.	Barium cmpds	secondary	1.64e-11	3.16e-10
Fuel Oil #4 Prod.	Chromium (III)	secondary	1.48e-11	2.85e-10
LPG Production	Halogenated matter (organic)	secondary	1.12e-11	2.16e-10
Fuel Oil #2 Prod.	Cadmium cmpds	secondary	9.02e-12	1.74e-10
Fuel Oil #6 Prod.	Chromium (VI)	secondary	8.55e-12	1.65e-10
Fuel Oil #4 Prod.	Lead cmpds	secondary	8.08e-12	1.56e-10
Fuel Oil #6 Prod.	Cadmium cmpds	secondary	7.93e-12	1.53e-10
Fuel Oil #4 Prod.	Toluene	secondary	7.52e-12	1.45e-10
Fuel Oil #2 Prod.	Chromium (VI)	secondary	6.91e-12	1.33e-10
Fuel Oil #6 Prod.	Chromium (III)	secondary	6.13e-12	1.18e-10
Natural Gas Prod.	Aromatic hydrocarbons	secondary	5.58e-12	1.08e-10
Fuel Oil #2 Prod.	Chromium (III)	secondary	4.96e-12	9.56e-11
Natural Gas Prod.	Barium cmpds	secondary	4.53e-12	8.73e-11
Fuel Oil #2 Prod.	Lead cmpds	secondary	3.01e-12	5.81e-11
Fuel Oil #2 Prod.	Toluene	secondary	2.80e-12	5.40e-11
Fuel Oil #6 Prod.	Lead cmpds	secondary	2.65e-12	5.11e-11
Fuel Oil #6 Prod.	Toluene	secondary	2.46e-12	4.75e-11
Natural Gas Prod.	Cadmium cmpds	secondary	2.19e-12	4.23e-11
Fuel Oil #4 Prod.	Nickel cmpds	secondary	9.78e-13	1.89e-11
Natural Gas Prod.	Lead cmpds	secondary	7.33e-13	1.41e-11
Natural Gas Prod.	Toluene	secondary	6.82e-13	1.31e-11
Fuel Oil #4 Prod.	Mercury compounds	secondary	4.39e-13	8.47e-12
Fuel Oil #2 Prod.	Nickel cmpds	secondary	3.65e-13	7.03e-12
Fuel Oil #6 Prod.	Nickel cmpds	secondary	3.21e-13	6.18e-12
Fuel Oil #4 Prod.	Cvanide (-1)	secondary	2.87e-13	5.54e-12
Fuel Oil #2 Prod.	Mercury compounds	secondary	1.64e-13	3.16e-12
Fuel Oil #6 Prod	Mercury compounds	secondary	1 44e-13	2.78e-12
Fuel Oil #2 Prod.	Cvanide (-1)	secondary	1.07e-13	2.06e-12
Fuel Oil #6 Prod	Cyanide (-1)	secondary	9.41e-14	1.81e-12
Natural Gas Prod	Nickel cmpds	secondary	8 87e-14	1.71e-12
Fuel Oil #4 Prod	Halogenated matter (organic)	secondary	8 20e-14	1 58e-12
Natural Gas Prod	Mercury compounds	secondary	3 98e-14	7.68e-13
Fuel Oil #2 Prod	Halogenated matter (organic)	secondary	3.06e-14	5 89e-13
Fuel Oil #6 Prod	Halogenated matter (organic)	secondary	2 69e-14	5 18e-13
Natural Gas Prod	Cvanide (-1)	secondary	2.60e-14	5.02e-13
Natural Gas Prod	Halogenated matter (organic)	secondary	7 44e-15	1 43e-13
Total Manufasturing	Theogenated matter (organic)	secondary	7.13	0.00 .01
i otal manufacturing			5.12e+00	9.886+01

Use, Maintenance and Repair Life-cycle Stage

Process Group	Material	LCI Data Type	Aquatic Toxicity (tox-kg) %	of Total
Total Use. Maintenance and R	enair	* *	0.00e+00	0.00e+00
End_of_life L ife_cycle Stage	-F		0.000	0.000
LCD landfilling	Ammonia	nrimory	1.860.05	3 580 04
LCD landfilling	Cilver compounds	primary	1.600-05	2.220.05
	Silver compounds	primary	2.0707	1.72 0.05
LCD incineration	Silver compounds	secondary	8.9/e-0/	6.762.07
LCD landfilling	Cadmium emplas	secondary	3.516-08	6.242.07
	Davium amp da	primary	2.0% 08	4.020.07
LCD incidentiation	Barium emple	secondary	2.086-08	4.020-07
LCD landming	Banun chipus	pi linai y	1.750-08	2 220 07
LPG Production		secondary	0.6% 00	1.872.07
	Aluminum (+3)	secondary	9.086-09	1.876-07
LCD incineration	A manife and the	secondary	8.086-09	1.50-07
LCD incineration	Arsenic cmpas	secondary	7.75e-09	1.500-07
	Mercury compounds	primary	6.99e-09	1.35e-07
LCD in sin susting	Arsenic cmpas	primary	6.59e-09	1.2/e-0/
LCD incineration	Lead cmpds	secondary	5.11e-09	9.85e-08
	Lead cmpds	primary	4.73e-09	9.12e-08
LCD incineration	Chromium (VI)	secondary	3.40e-09	6.55e-08
	Chromium (VI)	primary	2.72e-09	5.25e-08
LCD incineration	Chromium (III)	secondary	2.44e-09	4.70e-08
LCD landfilling	Chromium (III)	primary	1.95e-09	3.77e-08
LCD landfilling	Xylene (mixed isomers)	primary	1.01e-09	1.94e-08
LCD landfilling	Toluene	primary	5.68e-10	1.10e-08
LPG Production	Nitrate	secondary	4.77e-10	9.20e-09
LCD incineration	o-xylene	secondary	4.67e-10	9.01e-09
LCD incineration	Toluene	secondary	4.38e-10	8.45e-09
LPG Production	Fluorides (F-)	secondary	2.68e-10	5.17e-09
LCD landfilling	Ethylbenzene	primary	2.13e-10	4.12e-09
LCD incineration	Ethylbenzene	secondary	1.66e-10	3.19e-09
LPG Production	Copper (+1 & +2)	secondary	6.10e-11	1.18e-09
LCD landfilling	Tetrachloroethylene	primary	3.26e-11	6.28e-10
LCD incineration	Selenium	secondary	3.12e-11	6.02e-10
LCD landfilling	Selenium	primary	2.64e-11	5.09e-10
LCD incineration	Tetrachloroethylene	secondary	2.53e-11	4.87e-10
LCD landfilling	Benzene	primary	2.16e-11	4.17e-10
LPG Production	Zinc (+2)	secondary	1.71e-11	3.29e-10
LCD incineration	Benzene	secondary	1.68e-11	3.23e-10
LCD landfilling	Carbon tetrachloride	primary	1.32e-11	2.55e-10
LCD incineration	Carbon tetrachloride	secondary	1.03e-11	1.98e-10
LCD landfilling	Trichloroethylene	primary	9.99e-12	1.93e-10
LCD incineration	Trichloroethylene	secondary	7.75e-12	1.49e-10
LCD landfilling	Chloroform	primary	5.35e-12	1.03e-10
LCD landfilling	Vinyl chloride	primary	5.34e-12	1.03e-10
LCD incineration	Chloroform	secondary	4.15e-12	8.01e-11
LCD incineration	Vinyl chloride	secondary	4.14e-12	7.99e-11
LCD landfilling	1,2-Dichloroethane	primary	2.81e-12	5.42e-11
LCD incineration	1,2-Dichloroethane	secondary	2.18e-12	4.20e-11
LCD landfilling	Dichloromethane	primary	1.85e-12	3.57e-11
LCD incineration	Dichloromethane	secondary	1.44e-12	2.77e-11
LPG Production	Aromatic hydrocarbons	secondary	6.90e-13	1.33e-11

			Aquatic Toxicity	
Process Group	Material	LCI Data Type	(tox-kg)	% of Total
LPG Production	Barium cmpds	secondary	5.60e-13	1.08e-11
LPG Production	Chromium (VI)	secondary	4.00e-13	7.71e-12
LPG Production	Chromium (III)	secondary	2.87e-13	5.53e-12
LPG Production	Cadmium cmpds	secondary	2.71e-13	5.23e-12
LPG Production	Lead cmpds	secondary	9.06e-14	1.75e-12
LPG Production	Toluene	secondary	8.42e-14	1.62e-12
LPG Production	Nickel cmpds	secondary	1.10e-14	2.11e-13
LPG Production	Mercury compounds	secondary	4.93e-15	9.50e-14
LPG Production	Cyanide (-1)	secondary	3.22e-15	6.20e-14
LPG Production	Halogenated matter (organic)	secondary	9.19e-16	1.77e-14
Natural Gas Prod.	Halogenated matter (organic)	secondary	-1.56e-15	-3.01e-14
Natural Gas Prod.	Cyanide (-1)	secondary	-5.47e-15	-1.05e-13
Natural Gas Prod.	Mercury compounds	secondary	-8.37e-15	-1.61e-13
Natural Gas Prod.	Nickel cmpds	secondary	-1.86e-14	-3.59e-13
LCD incineration	Halogenated matter (organic)	secondary	-2.38e-14	-4.59e-13
LCD incineration	Cyanide (-1)	secondary	-8.33e-14	-1.61e-12
Natural Gas Prod.	Toluene	secondary	-1.43e-13	-2.76e-12
Natural Gas Prod.	Lead cmpds	secondary	-1.54e-13	-2.97e-12
LCD incineration	Nickel cmpds	secondary	-2.84e-13	-5.47e-12
Fuel Oil #4 Prod.	Halogenated matter (organic)	secondary	-3.78e-13	-7.29e-12
Natural Gas Prod	Cadmium cmpds	secondary	-4 61e-13	-8 88e-12
Natural Gas Prod	Barium cmpds	secondary	-9 51e-13	-1 83e-11
Natural Gas Prod	Aromatic hydrocarbons	secondary	-1 17e-12	-2 26e-11
Fuel Oil #4 Prod	Cyanide (-1)	secondary	-1 32e-12	-2 55e-11
Fuel Oil #4 Prod	Mercury compounds	secondary	-2.02e-12	-3 90e-11
Fuel Oil #4 Prod	Nickel compounds	secondary	4 510 12	-5.50c-11
I CD incineration	A romatic hydrocathons	secondary	-1.78e-11	-3.44e-10
Evol Oil #4 Prod	Teluene	secondary	-1./80-11	-5.440-10
Fuel Oil #4 Prod	Load ampda	secondary	-3.406-11	-0.086-10
Fuel Oil #4 Prod	Chromium (III)	secondary	-3.720-11	-7.180-10
Fuel Oil #4 Prod	Chromium (III)	secondary	-0.010-11	-1.316-09
Natural Cas Brad	Carrow (+1, 8, +2)	secondary	-9.306-11	-1.836-09
Natural Gas Prod.	Copper $(+1 \alpha + 2)$	secondary	-1.04e-10	-2.00e-09
Fuel Oil #4 Prod.	Cadmium empos	secondary	-1.11e-10	-2.15e-09
Fuel Oil #4 Prod.	Barium cmpds	secondary	-2.30e-10	-4.44e-09
Fuel Oil #4 Prod.	Aromatic hydrocarbons	secondary	-2.83e-10	-5.4/e-09
Natural Gas Prod.	Chromium (III)	secondary	-7.28e-10	-1.40e-08
Natural Gas Prod.	Chromium (VI)	secondary	-1.01e-09	-1.96e-08
LCD incineration	Copper $(+1 \& +2)$	secondary	-1.58e-09	-3.04e-08
LCD incineration	Zinc (+2)	secondary	-5.36e-09	-1.03e-07
Fuel Oil #4 Prod.	Zinc (+2)	secondary	-7.83e-09	-1.51e-07
Natural Gas Prod.	Nitrate	secondary	-9.96e-09	-1.92e-07
Natural Gas Prod.	Aluminum (+3)	secondary	-1.65e-08	-3.17e-07
Fuel Oil #4 Prod.	Copper (+1 & +2)	secondary	-2.51e-08	-4.83e-07
Natural Gas Prod.	Phenol	secondary	-2.86e-08	-5.52e-07
Natural Gas Prod.	Fluorides (F-)	secondary	-3.95e-08	-7.61e-07
Natural Gas Prod.	Zinc (+2)	secondary	-6.41e-08	-1.24e-06
Fuel Oil #4 Prod.	Fluorides (F-)	secondary	-1.62e-07	-3.12e-06
Fuel Oil #4 Prod.	Nitrate	secondary	-2.08e-07	-4.02e-06
LCD incineration	Aluminum (+3)	secondary	-2.51e-07	-4.83e-06

			Aquatic Toxicity	
Process Group	Material	LCI Data Type	(tox-kg)	% of Total
LCD incineration	Phenol	secondary	-4.33e-07	-8.36e-06
LCD incineration	Nitrate	secondary	-2.33e-06	-4.49e-05
Fuel Oil #4 Prod.	Aluminum (+3)	secondary	-3.98e-06	-7.68e-05
Fuel Oil #4 Prod.	Phenol	secondary	-6.88e-06	-1.33e-04
LCD incineration	Fluorides (F-)	secondary	-9.81e-06	-1.89e-04
LCD incineration	Ammonia	secondary	-2.34e-04	-4.51e-03
Total End-of-life			-2.37e-04	-4.58e-03
Total All Life-cycle Stages			5.19e+00	1.00e+02

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Materials Processing Life-cycle S	tage			
Invar	Sulfur dioxide	secondary	1.65e+02	8.35e+00
Steel Prod., cold-rolled, semi-finished	Sulfur dioxide	secondary	2.54e+01	1.28e+00
Aluminum Prod.	Sulfur dioxide	secondary	1.38e+01	6.99e-01
Polycarbonate Production	Sulfur dioxide	secondary	7.93e+00	4.02e-01
Lead	Sulfur dioxide	secondary	4.57e+00	2.32e-01
ABS Production	Sulfur dioxide	secondary	2.80e+00	1.42e-01
Ferrite mfg.	Sulfur dioxide	secondary	2.35e+00	1.19e-01
Lead	Arsenic	secondary	2.21e+00	1.12e-01
Polystyrene Prod., high-impact	Sulfur dioxide	secondary	1.20e+00	6.07e-02
Aluminum Prod.	Titanium tetrachloride	secondary	7.20e-01	3.65e-02
Steel Prod., cold-rolled, semi-finished	Carbon monoxide	secondary	2.78e-01	1.41e-02
Invar	Titanium tetrachloride	secondary	2.56e-01	1.30e-02
Lead	Titanium tetrachloride	secondary	2.10e-01	1.07e-02
Aluminum Prod.	Manganese cmpds	secondary	1.62e-01	8.20e-03
Invar	Manganese cmpds	secondary	1.50e-01	7.58e-03
Aluminum Prod.	Vanadium	secondary	1.07e-01	5.44e-03
Steel Prod., cold-rolled, semi-finished	PM	secondary	1.05e-01	5.31e-03
Ferrite mfg.	Manganese cmpds	secondary	9.39e-02	4.76e-03
Invar	Vanadium	secondary	7.40e-02	3.75e-03
Invar	Carbon monoxide	secondary	7.04e-02	3.56e-03
Ferrite mfg.	Carbon monoxide	secondary	5.85e-02	2.96e-03
Steel Prod., cold-rolled, semi-finished	Vanadium	secondary	5.15e-02	2.61e-03
Aluminum Prod.	Arsenic cmpds	secondary	4.75e-02	2.41e-03
Lead	Manganese cmpds	secondary	4.60e-02	2.33e-03
Polycarbonate Production	Carbon monoxide	secondary	4.15e-02	2.10e-03
Invar	Arsenic	secondary	3.21e-02	1.62e-03
Ferrite mfg.	Arsenic	secondary	2.90e-02	1.47e-03
Lead	Zinc (elemental)	secondary	2.42e-02	1.23e-03
Ferrite mfg.	Titanium tetrachloride	secondary	2.36e-02	1.20e-03
Invar	Zinc (elemental)	secondary	2.28e-02	1.15e-03
Ferrite mfg.	Zinc (elemental)	secondary	2.21e-02	1.12e-03
Lead	Vanadium	secondary	2.09e-02	1.06e-03
Polycarbonate Production	Methane	secondary	2.03e-02	1.03e-03
ABS Production	Carbon monoxide	secondary	2.01e-02	1.02e-03
Polycarbonate Production	Nitrogen dioxide	secondary	1.94e-02	9.82e-04
Steel Prod., cold-rolled, semi-finished	Fluorides (F-)	secondary	1.91e-02	9.68e-04
Lead	Carbon monoxide	secondary	1.89e-02	9.57e-04
Aluminum Prod.	Carbon monoxide	secondary	1.72e-02	8.74e-04
Invar	Arsenic cmpds	secondary	1.69e-02	8.55e-04
Styrene-butadiene Copolymer Prod.	Carbon monoxide	secondary	1.60e-02	8.12e-04
Steel Prod., cold-rolled, semi-finished	Manganese cmpds	secondary	1.49e-02	7.55e-04
Ferrite mfg.	Vanadium	secondary	1.45e-02	7.35e-04
Steel Prod., cold-rolled, semi-finished	Nitrogen dioxide	secondary	1.40e-02	7.09e-04
Lead	Arsenic cmpds	secondary	1.38e-02	6.99e-04
Steel Prod., cold-rolled, semi-finished	Methane	secondary	1.09e-02	5.52e-04
Aluminum Prod.	Barium cmpds	secondary	1.04e-02	5.29e-04
Aluminum Prod.	PM	secondary	8.90e-03	4.51e-04
Aluminum Prod.	Methane	secondary	8.05e-03	4.08e-04
Aluminum Prod.	Nitrogen dioxide	secondary	7.13e-03	3.61e-04

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Styrene-butadiene Copolymer Prod.	Nitrogen oxides	secondary	6.99e-03	3.54e-04
Styrene-butadiene Copolymer Prod.	Methane	secondary	6.87e-03	3.48e-04
Polycarbonate Production	PM	secondary	6.46e-03	3.27e-04
Aluminum Prod.	Selenium	secondary	6.29e-03	3.19e-04
Invar	Methane	secondary	5.76e-03	2.92e-04
Styrene-butadiene Copolymer Prod.	Sulfur oxides	secondary	5.71e-03	2.89e-04
ABS Production	Methane	secondary	5.08e-03	2.57e-04
Invar	Nitrogen dioxide	secondary	5.07e-03	2.57e-04
ABS Production	Nitrogen dioxide	secondary	4.66e-03	2.36e-04
Steel Prod., cold-rolled, semi-finished	Arsenic	secondary	4.52e-03	2.29e-04
Invar	Barium cmpds	secondary	4.31e-03	2.19e-04
Lead	Nitrogen dioxide	secondary	4.18e-03	2.12e-04
Steel Prod., cold-rolled, semi-finished	Phosphorus (yellow or white)	secondary	4.07e-03	2.06e-04
ABS Production	Hydrochloric acid	secondary	2.93e-03	1.48e-04
Ferrite mfg.	Methane	secondary	2.85e-03	1.45e-04
Invar	PM	secondary	2.76e-03	1.40e-04
Lead	Barium cmpds	secondary	2.71e-03	1.37e-04
Lead	Methane	secondary	2.52e-03	1.28e-04
Invar	Selenium	secondary	2.32e-03	1.18e-04
Polystyrene Prod., high-impact	Carbon monoxide	secondary	2.27e-03	1.15e-04
Steel Prod., cold-rolled, semi-finished	Hydrochloric acid	secondary	2.05e-03	1.04e-04
Ferrite mfg.	Nitrogen dioxide	secondary	2.03e-03	1.03e-04
Invar	Hydrochloric acid	secondary	1.85e-03	9.38e-05
Lead	Selenium	secondary	1.84e-03	9.30e-05
Polystyrene Prod., high-impact	Nitrogen dioxide	secondary	1.81e-03	9.19e-05
Aluminum Prod.	Hydrochloric acid	secondary	1.78e-03	9.04e-05
Steel Prod., cold-rolled, semi-finished	Benzene	secondary	1.76e-03	8.91e-05
Steel Prod., cold-rolled, semi-finished	Barium cmpds	secondary	1.73e-03	8.77e-05
Ferrite mfg.	PM	secondary	1.72e-03	8.72e-05
Polystyrene Prod., high-impact	Methane	secondary	1.66e-03	8.42e-05
Ferrite mfg.	Arsenic cmpds	secondary	1.64e-03	8.30e-05
ABS Production	Formaldehyde	secondary	1.60e-03	8.11e-05
Aluminum Prod.	Aluminum (+3)	secondary	1.59e-03	8.04e-05
Lead	Lead	secondary	1.58e-03	8.01e-05
Steel Prod., cold-rolled, semi-finished	Arsenic cmpds	secondary	1.54e-03	7.78e-05
ABS Production	PM	secondary	1.27e-03	6.44e-05
Aluminum Prod.	Cadmium cmpds	secondary	1.17e-03	5.93e-05
Lead	Hydrochloric acid	secondary	1.07e-03	5.42e-05
Styrene-butadiene Copolymer Prod.	PM	secondary	9.93e-04	5.03e-05
Invar	Copper	secondary	9.89e-04	5.01e-05
Invar	Benzene	secondary	9.17e-04	4.65e-05
Lead	PM	secondary	8.41e-04	4.26e-05
Steel Prod., cold-rolled, semi-finished	Titanium tetrachloride	secondary	8.33e-04	4.22e-05
Steel Prod., cold-rolled, semi-finished	Ammonia	secondary	7.99e-04	4.05e-05
Ferrite mfg.	Barium cmpds	secondary	7.81e-04	3.96e-05
Steel Prod., cold-rolled, semi-finished	Nitrous oxide	secondary	7.77e-04	3.94e-05
Ferrite mfg.	Hydrochloric acid	secondary	6.93e-04	3.51e-05
Ferrite mfg.	Benzene	secondary	5.85e-04	2.97e-05
Invar	Aluminum (+3)	secondary	5.70e-04	2.89e-05
Aluminum Prod.	Benzene	secondary	5.69e-04	2.88e-05

Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Steel Prod., cold-rolled, semi-finished	Silicon	secondary	5.06e-04	2.56e-05
Polycarbonate Production	Hydrochloric acid	secondary	4.65e-04	2.36e-05
Lead	Aluminum (+3)	secondary	4.62e-04	2.34e-05
Aluminum Prod.	Barium sulfate	secondary	4.54e-04	2.30e-05
Steel Prod., cold-rolled, semi-finished	Formaldehyde	secondary	4.15e-04	2.10e-05
Steel Prod., cold-rolled, semi-finished	Titanium	secondary	3.26e-04	1.65e-05
Polycarbonate Production	Sulfuric acid	secondary	3.17e-04	1.61e-05
Polystyrene Prod., high-impact	PM	secondary	3.02e-04	1.53e-05
Steel Prod., cold-rolled, semi-finished	Selenium	secondary	3.00e-04	1.52e-05
Lead	Benzene	secondary	2.99e-04	1.51e-05
Invar	Phosphorus (yellow or white)	secondary	2.89e-04	1.46e-05
Ferrite mfg.	Selenium	secondary	2.86e-04	1.45e-05
Ferrite mfg.	Phosphorus (yellow or white)	secondary	2.85e-04	1.44e-05
Steel Prod., cold-rolled, semi-finished	Ethane	secondary	2.46e-04	1.25e-05
ABS Production	Hydrofluoric acid	secondary	2.44e-04	1.23e-05
Polycarbonate Production	Mercury compounds	secondary	2.43e-04	1.23e-05
Aluminum Prod.	Hydrofluoric acid	secondary	2.40e-04	1.22e-05
Aluminum Prod.	Zinc (elemental)	secondary	2.30e-04	1.16e-05
Aluminum Prod.	Titanium	secondary	2.28e-04	1.15e-05
Invar	Silicon	secondary	2.26e-04	1.15e-05
Steel Prod., cold-rolled, semi-finished	Zinc (elemental)	secondary	2.16e-04	1.10e-05
Invar	Titanium	secondary	2.13e-04	1.08e-05
ABS Production	Ammonia	secondary	1.94e-04	9.82e-06
ABS Production	Aromatic hydrocarbons	secondary	1.91e-04	9.65e-06
Steel Prod., cold-rolled, semi-finished	Molybdenum	secondary	1.88e-04	9.51e-06
Aluminum Prod.	Copper (+1 & +2)	secondary	1.80e-04	9.12e-06
Invar	Formaldehyde	secondary	1.60e-04	8.11e-06
Ferrite mfg.	Formaldehyde	secondary	1.56e-04	7.92e-06
Invar	Molybdenum	secondary	1.52e-04	7.70e-06
Ferrite mfg.	Silicon	secondary	1.48e-04	7.52e-06
ABS Production	Sulfuric acid	secondary	1.45e-04	7.37e-06
Styrene-butadiene Copolymer Prod.	Sulfuric acid	secondary	1.42e-04	7.20e-06
Aluminum Prod.	Nitrous oxide	secondary	1.38e-04	7.00e-06
Aluminum Prod.	Perfluoromethane	secondary	1.30e-04	6.57e-06
Invar	Cadmium cmpds	secondary	1.27e-04	6.43e-06
Steel Prod., cold-rolled, semi-finished	Barium	secondary	1.23e-04	6.23e-06
Invar	Nitrates/nitrites	secondary	1.21e-04	6.11e-06
ABS Production	Ethane	secondary	1.19e-04	6.01e-06
Lead	Barium sulfate	secondary	1.15e-04	5.81e-06
ABS Production	Mercury compounds	secondary	1.11e-04	5.65e-06
Aluminum Prod.	Silicon	secondary	1.09e-04	5.54e-06
Styrene-butadiene Copolymer Prod.	Mercury compounds	secondary	1.09e-04	5.52e-06
Steel Prod., cold-rolled, semi-finished	Tin (Sn++, Sn4+)	secondary	1.04e-04	5.28e-06
Steel Prod., cold-rolled, semi-finished	Antimony	secondary	9.51e-05	4.82e-06
Invar	Copper (+1 & +2)	secondary	9.43e-05	4.78e-06
Polycarbonate Production	Fluorides (F-)	secondary	9.16e-05	4.64e-06
Ferrite mfg.	Titanium	secondary	9.12e-05	4.62e-06
Invar	Ethane	secondary	8.26e-05	4.18e-06
Polycarbonate Production	Aromatic hydrocarbons	secondary	8.22e-05	4.16e-06
Ferrite mfg.	Ethane	secondary	8.06e-05	4.09e-06

Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Lead	Silicon	secondary	7.13e-05	3.61e-06
Styrene-butadiene Copolymer Prod.	Hydrochloric acid	secondary	6.82e-05	3.46e-06
Styrene-butadiene Copolymer Prod.	Aromatic hydrocarbons	secondary	6.79e-05	3.44e-06
Invar	Nickel	secondary	6.54e-05	3.31e-06
Invar	Barium	secondary	6.53e-05	3.31e-06
Steel Prod., cold-rolled, semi-finished	Silver compounds	secondary	6.17e-05	3.13e-06
Ferrite mfg.	Aluminum (+3)	secondary	5.58e-05	2.82e-06
Ferrite mfg.	Molybdenum	secondary	5.23e-05	2.65e-06
Lead	Copper (+1 & +2)	secondary	5.20e-05	2.64e-06
ABS Production	Aluminum (+3)	secondary	5.08e-05	2.57e-06
Steel Prod., cold-rolled, semi-finished	Nitrates/nitrites	secondary	4.99e-05	2.53e-06
Steel Prod., cold-rolled, semi-finished	Boron	secondary	4.93e-05	2.50e-06
Aluminum Prod.	Nitrate	secondary	4.56e-05	2.31e-06
Steel Prod., cold-rolled, semi-finished	Mercury compounds	secondary	4.52e-05	2.29e-06
Styrene-butadiene Copolymer Prod.	Fluorides (F-)	secondary	4.10e-05	2.08e-06
Aluminum Prod.	Barium	secondary	3.91e-05	1.98e-06
Ferrite mfg.	Barium	secondary	3.73e-05	1.89e-06
Invar	Lead	secondary	3.62e-05	1.83e-06
Ferrite mfg.	Lead	secondary	3.46e-05	1.75e-06
Invar	Barium sulfate	secondary	3.46e-05	1.75e-06
Steel Prod., cold-rolled, semi-finished	Barium sulfate	secondary	3.43e-05	1.74e-06
Ferrite mfg.	Barium sulfate	secondary	3.38e-05	1.71e-06
Invar	Ammonia	secondary	3.16e-05	1.60e-06
Lead	Ethane	secondary	3.16e-05	1.60e-06
Invar	Tin (Sn++, Sn4+)	secondary	3.04e-05	1.54e-06
Polystyrene Prod., high-impact	Aromatic hydrocarbons	secondary	3.02e-05	1.53e-06
ABS Production	Nitrate	secondary	3.01e-05	1.52e-06
Lead	Barium	secondary	2.88e-05	1.46e-06
Lead	Copper	secondary	2.82e-05	1.43e-06
Invar	Boron	secondary	2.79e-05	1.41e-06
Steel Prod., cold-rolled, semi-finished	Cadmium cmpds	secondary	2.70e-05	1.37e-06
Invar	Antimony	secondary	2.69e-05	1.37e-06
Invar	Hydrofluoric acid	secondary	2.69e-05	1.36e-06
Ferrite mfg.	Antimony	secondary	2.63e-05	1.33e-06
Steel Prod., cold-rolled, semi-finished	Aluminum (elemental)	secondary	2.47e-05	1.25e-06
Invar	Zinc (+2)	secondary	2.44e-05	1.23e-06
Polystyrene Prod., high-impact	Hydrochloric acid	secondary	2.42e-05	1.23e-06
Aluminum Prod.	Copper	secondary	2.38e-05	1.21e-06
Steel Prod., cold-rolled, semi-finished	Acetylene	secondary	2.33e-05	1.18e-06
Lead	Nitrous oxide	secondary	2.03e-05	1.03e-06
Aluminum Prod.	Nickel cmpds	secondary	1.91e-05	9.65e-07
Steel Prod., cold-rolled, semi-finished	Hydrofluoric acid	secondary	1.89e-05	9.59e-07
Invar	Silver compounds	secondary	1.86e-05	9.44e-07
Ferrite mfg.	Zinc (+2)	secondary	1.84e-05	9.34e-07
Ferrite mfg.	Silver compounds	secondary	1.82e-05	9.22e-07
Aluminum Prod.	Nickel	secondary	1.78e-05	9.00e-07
Aluminum Prod.	Zinc (+2)	secondary	1.74e-05	8.83e-07
Steel Prod., cold-rolled, semi-finished	Pentane	secondary	1.70e-05	8.59e-07
Steel Prod., cold-rolled, semi-finished	Aluminum (+3)	secondary	1.58e-05	7.98e-07
Aluminum Prod.	Ammonia	secondary	1.55e-05	7.84e-07
		~		

Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Invar	Aluminum (elemental)	secondary	1.52e-05	7.72e-07
Invar	Nickel cmpds	secondary	1.51e-05	7.66e-07
Ferrite mfg.	Boron	secondary	1.48e-05	7.51e-07
Aluminum Prod.	Aromatic hydrocarbons	secondary	1.41e-05	7.17e-07
Lead	Hydrofluoric acid	secondary	1.36e-05	6.91e-07
Steel Prod., cold-rolled, semi-finished	Hydrogen sulfide	secondary	1.36e-05	6.88e-07
Aluminum Prod.	Nitrites	secondary	1.35e-05	6.85e-07
Steel Prod., cold-rolled, semi-finished	Copper	secondary	1.22e-05	6.19e-07
Lead	Ammonia	secondary	1.19e-05	6.01e-07
Ferrite mfg.	Nitrous oxide	secondary	1.06e-05	5.38e-07
Aluminum Prod.	Aluminum (elemental)	secondary	1.05e-05	5.33e-07
Polycarbonate Production	Copper (+1 & +2)	secondary	1.04e-05	5.25e-07
Ferrite mfg.	Ammonia	secondary	1.03e-05	5.19e-07
ABS Production	Hydrogen sulfide	secondary	1.01e-05	5.13e-07
Aluminum Prod.	Strontium (Sr II)	secondary	1.00e-05	5.06e-07
Lead	Boron	secondary	9.53e-06	4.83e-07
Polycarbonate Production	Phenol	secondary	9.34e-06	4.73e-07
Aluminum Prod	Fluoride	secondary	9 10e-06	4 61e-07
Lead	Nitrate	secondary	9.00e-06	4 56e-07
Invar	Pentane	secondary	8 97e-06	4 55e-07
Ferrite mfg	Pentane	secondary	8.76e-06	4 44e-07
Aluminum Prod	Lead cmpds	secondary	8.70e-06	4 42e-07
Steel Prod cold-rolled semi-finished	Copper $(+1 & +2)$	secondary	8.59e-06	4 35e-07
Invar	Aromatic hydrocarbons	secondary	8.36e-06	4.33e-07
Lead	Aluminum (elemental)	secondary	8.16e-06	4 13e-07
Steel Prod cold-rolled semi-finished	Nickel	secondary	8.14e-06	4.13e-07
Aluminum Prod	Vanadium (V3+ V5+)	secondary	8.14c-00	4.13e-07
Ferrite mfg	Aluminum (elemental)	secondary	7.77e.06	3.03e.07
Invar	Hydrogen sulfide	secondary	7.77C=00	3.55C=07
Steel Brod cold rolled semi finished	A romatic hydrocarbons	secondary	7.200-00	3.63e.07
Inver	Acetylene	secondary	6.610.06	3.350.07
Forrito mfa	Acetylene	secondary	6.460.06	3.336-07
Ferrite mfg.	Acetylene Uvdrafluaria aaid	secondary	6.270.06	3.276-07
Ferrite mfg.		secondary	5.0(00	3.23e-07
Ferrite inig.	Nitwitter	secondary	5.966-06	3.02e-07
Invar Equite m fo	Codmium omn do	secondary	5.956-06	3.01e-07
Fernie inig.	Nieleel energie	secondary	5.946-00	3.01e-07
Lead	Nickel emplos	secondary	5.000-00	2.836-07
Invar	Strontium (Sr II)	secondary	5.18e-06	2.62e-07
Ferrite mig.	Hydrogen sulfide	secondary	5.17e-06	2.62e-07
ABS Production	Heptane	secondary	5.02e-06	2.54e-07
Lead	Zinc (+2)	secondary	4.96e-06	2.51e-07
Aluminum Prod.	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI)	secondary	4.94e-06	2.50e-07
ABS Production	Copper (+1 & +2)	secondary	4.75e-06	2.41e-07
Lead	Nickel	secondary	4.72e-06	2.39e-07
Steel Prod cold-rolled semi-finished	Strontium (Sr II)	secondary	4 67e-06	2.37e-07
Styrene-butadiene Conolymer Prod	Copper $(+1 \& +2)$	secondary	4.64e-06	2.37e-07
Ferrite mfg	Copper	secondary	4 55e-06	2.31e-07
Lead	Fluoride	secondary	4.46e-06	2.510 07
Steel Prod cold-rolled semi-finished	Hexane	secondary	4 43e-06	2.200 07 2.24e-07
see in the second secon		Secondary		2.270 07

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category
Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Invar	Fluoride	secondary	4.40e-06	2.23e-07
Ferrite mfg.	Nitrate	secondary	4.04e-06	2.05e-07
Lead	Cadmium	secondary	3.98e-06	2.02e-07
Aluminum Prod.	Hydrogen sulfide	secondary	3.77e-06	1.91e-07
Polycarbonate Production	Ammonia	secondary	3.70e-06	1.87e-07
Polycarbonate Production	Hydrofluoric acid	secondary	3.69e-06	1.87e-07
Steel Prod., cold-rolled, semi-finished	Chromium (VI)	secondary	3.30e-06	1.67e-07
Invar	Vanadium (V3+, V5+)	secondary	3.00e-06	1.52e-07
Invar	Lead cmpds	secondary	2.98e-06	1.51e-07
Lead	Hydrogen sulfide	secondary	2.96e-06	1.50e-07
Styrene-butadiene Copolymer Prod.	Nitrate	secondary	2.90e-06	1.47e-07
Lead	Nitrites	secondary	2.86e-06	1.45e-07
Lead	Lead cmpds	secondary	2.75e-06	1.39e-07
Lead	Vanadium (V3+, V5+)	secondary	2.37e-06	1.20e-07
Ferrite mfg.	Nickel	secondary	2.30e-06	1.16e-07
Ferrite mfg.	Aromatic hydrocarbons	secondary	2.26e-06	1.15e-07
Steel Prod., cold-rolled, semi-finished	Boric acid	secondary	2.17e-06	1.10e-07
Steel Prod., cold-rolled, semi-finished	Bromine	secondary	2.03e-06	1.03e-07
Lead	Strontium (Sr II)	secondary	2.00e-06	1.01e-07
Invar	Methanol	secondary	1.87e-06	9.45e-08
Polycarbonate Production	Halogenated hydrocarbons (unspecified)	secondary	1.85e-06	9.36e-08
Polycarbonate Production	Hydrogen sulfide	secondary	1.85e-06	9.36e-08
Aluminum Prod.	Toluene	secondary	1.81e-06	9.18e-08
Invar	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI)	secondary	1.72e-06	8.73e-08
Steel Prod., cold-rolled, semi-finished	Uranium	secondary	1.68e-06	8.53e-08
Polystyrene Prod., high-impact	Ammonia	secondary	1.62e-06	8.18e-08
Steel Prod., cold-rolled, semi-finished	Toluene	secondary	1.52e-06	7.69e-08
Ferrite mfg.	Strontium (Sr II)	secondary	1.52e-06	7.68e-08
Lead	Mercury	secondary	1.51e-06	7.66e-08
Lead	Molybdenum (Mo II, Mo III, Mo IV, Mo V, Mo VI)	secondary	1.49e-06	7.57e-08
Invar	Cobalt (Co I, Co II, Co III)	secondary	1.44e-06	7.31e-08
Aluminum Prod.	Xylene (mixed isomers)	secondary	1.38e-06	6.98e-08
Invar	Hexane	secondary	1.29e-06	6.52e-08
Ferrite mfg.	Hexane	secondary	1.26e-06	6.37e-08
Invar	Boron (B III)	secondary	1.26e-06	6.36e-08
Invar	Benzo[a]pyrene	secondary	1.23e-06	6.23e-08
Steel Prod., cold-rolled, semi-finished	Ethylene	secondary	1.23e-06	6.22e-08
Steel Prod., cold-rolled, semi-finished	Cyanide (-1)	secondary	1.21e-06	6.11e-08
Aluminum Prod.	Lead	secondary	1.20e-06	6.08e-08
Steel Prod., cold-rolled, semi-finished	Methanol	secondary	1.19e-06	6.04e-08
Invar	Cadmium	secondary	1.18e-06	5.97e-08
Steel Prod., cold-rolled, semi-finished	Xylene (mixed isomers)	secondary	1.12e-06	5.68e-08
Steel Prod., cold-rolled, semi-finished	Lead	secondary	1.10e-06	5.58e-08
Steel Prod., cold-rolled, semi-finished	Naphthalene	secondary	1.10e-06	5.57e-08
Invar	Toluene	secondary	1.10e-06	5.57e-08
Lead	Bromine	secondary	1.08e-06	5.46e-08
Ferrite mfg.	Cadmium	secondary	9.76e-07	4.95e-08
Steel Prod., cold-rolled, semi-finished	Phenol	secondary	9.70e-07	4.91e-08
Steel Prod., cold-rolled, semi-finished	Heptane	secondary	9.33e-07	4.73e-08

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Invar	Chromium (VI)	secondary	9.25e-07	4.69e-08
Polycarbonate Production	Chlorine	secondary	9.23e-07	4.68e-08
Ferrite mfg.	Fluoride	secondary	9.23e-07	4.68e-08
Lead	Cobalt (Co I, Co II, Co III)	secondary	9.18e-07	4.65e-08
Aluminum Prod.	Chromium (VI)	secondary	8.90e-07	4.51e-08
Steel Prod., cold-rolled, semi-finished	Lead cmpds	secondary	8.83e-07	4.47e-08
Lead	Manganese	secondary	8.47e-07	4.29e-08
Styrene-butadiene Copolymer Prod.	Aluminum (+3)	secondary	8.27e-07	4.19e-08
Ferrite mfg.	Nickel cmpds	secondary	7.97e-07	4.04e-08
Lead	Mercury compounds	secondary	7.88e-07	3.99e-08
Invar	Xylene (mixed isomers)	secondary	7.67e-07	3.89e-08
Lead	Boron (B III)	secondary	7.54e-07	3.82e-08
Invar	Bromium (Br)	secondary	7.09e-07	3.59e-08
Ferrite mfg.	Bromine	secondary	6.92e-07	3.50e-08
Aluminum Prod.	Phenol	secondary	6.39e-07	3.24e-08
ABS Production	Chlorine	secondary	6.35e-07	3.22e-08
Invar	Manganese	secondary	6.32e-07	3.20e-08
Invar	Boric acid	secondary	6.13e-07	3.11e-08
Invar	Ethylene	secondary	6.01e-07	3.04e-08
Ferrite mfg.	Boric acid	secondary	5.99e-07	3.03e-08
Steel Prod., cold-rolled, semi-finished	Nickel cmpds	secondary	5.96e-07	3.02e-08
ABS Production	Phenol	secondary	5.88e-07	2.98e-08
Ferrite mfg.	Ethylene	secondary	5.79e-07	2.93e-08
Ferrite mfg.	Toluene	secondary	5.31e-07	2.69e-08
Steel Prod., cold-rolled, semi-finished	Zinc (+2)	secondary	5.30e-07	2.69e-08
Ferrite mfg.	Lead cmpds	secondary	5.19e-07	2.63e-08
ABS Production	Nickel cmpds	secondary	5.04e-07	2.55e-08
Styrene-butadiene Copolymer Prod.	Phenol	secondary	4.92e-07	2.49e-08
Invar	Uranium	secondary	4.81e-07	2.44e-08
Ferrite mfg.	Uranium	secondary	4.69e-07	2.38e-08
Polycarbonate Production	Aluminum (+3)	secondary	4.62e-07	2.34e-08
Polycarbonate Production	Ethanethiol	secondary	4.62e-07	2.34e-08
Polycarbonate Production	Lead	secondary	4.62e-07	2.34e-08
Polycarbonate Production	Mercury	secondary	4.62e-07	2.34e-08
Polycarbonate Production	Nickel (+2)	secondary	4.62e-07	2.34e-08
Polycarbonate Production	Nitrate	secondary	4.62e-07	2.34e-08
Polycarbonate Production	Nitrous oxide	secondary	4.62e-07	2.34e-08
Polycarbonate Production	Zinc (+2)	secondary	4.62e-07	2.34e-08
Invar	Phenol	secondary	4.41e-07	2.23e-08
Aluminum Prod.	Acetic acid	secondary	4.22e-07	2.14e-08
Ferrite mfg.	Chromium (VI)	secondary	4.17e-07	2.11e-08
Styrene-butadiene Copolymer Prod.	Chlorine	secondary	4.14e-07	2.10e-08
Styrene-butadiene Copolymer Prod.	Hydrofluoric acid	secondary	4.14e-07	2.10e-08
Ferrite mfg.	Nitrites	secondary	4.13e-07	2.09e-08
Aluminum Prod.	Cobalt	secondary	4.11e-07	2.08e-08
Steel Prod., cold-rolled, semi-finished	Cobalt	secondary	4.05e-07	2.05e-08
Lead	Chromium (VI)	secondary	3.70e-07	1.88e-08
Steel Prod., cold-rolled, semi-finished	Manganese	secondary	3.60e-07	1.82e-08
Styrene-butadiene Copolymer Prod.	Ammonia	secondary	3.55e-07	1.80e-08
Ferrite mfg.	Xylene (mixed isomers)	secondary	3.50e-07	1.77e-08

InvarAccil acidsecondary34.0-071.73-0.88Ferrin mig.Vanafur (V3+, V5-)secondary34.0-071.73-0.88Polystyrene Pred, high-inpactNitratesecondary2.05-071.53-0.85Stel Pred, chid-rolld, semi-finishVanafurn (V3+, V5-)secondary2.07-071.53-0.85Aluminm Prod.Cadmiansecondary2.28-071.53-0.85InvarCyanide (1)secondary2.28-071.46-0.85InvarHeptanesecondary2.76-071.46-0.85Stel Prod, chid-child, semi-finishedBroon (B III)secondary2.76-071.36-0.85InvarHeptanesecondary2.76-071.36-0.85InvarAccil aldoly desecondary2.76-071.36-0.85InvarAccil aldoly desecondary2.26-071.36-0.85ABS ProductionHAI ON-1301secondary2.24-071.36-0.85InvarMercurysecondary2.12-071.07-0.85ABS ProductionFintuschilonsecondary2.12-071.07-0.85ABS ProductionFintuschilonsecondary2.12-071.07-0.85ABS ProductionInadsecondary2.12-071.07-0.85ABS ProductionInadesecondary2.12-071.07-0.85ABS ProductionInadesecondary2.12-071.07-0.85ABS ProductionInadesecondary2.12-071.07-0.85ABS ProductionInadesecondary2.12-071.07-0.85 <th>Process Group</th> <th>Material</th> <th>LCI Data Type</th> <th>Terrestrial Toxicity (tox-kg)</th> <th>% of Total</th>	Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Ferrite nifg. Vanadiam (V3+, V3+) secondary 3.42e-07 1.73e-08 Peritre nifg. Mehanol secondary 3.02e-07 1.75e-08 Polytrone Prod., high-inpart Nitrate secondary 2.97e-07 1.50e-08 Aluminum Prod. Cadmiun secondary 2.97e-07 1.50e-08 Aluminum Prod. Cadmiun secondary 2.89e-07 1.45e-08 Invar Cyanide (-1) secondary 2.89e-07 1.45e-08 Forrite nifg. Berone OR II) secondary 2.76e-07 1.34e-08 Scele Prod., cold-rolled, semi-finided Rort OR II) secondary 2.24e-07 1.34e-08 Lad Nature OR III) secondary 2.12e-07 1.37e-08 Alas Production Halcon+1301 secondary 2.12e-07 1.07e-08 Alas Production Fluorard secondary 2.12e-07 1.07e-08 ABS Production Halogenated hydrocarbons (unspecified) secondary 2.12e-07 1.07e-08 ABS Production Zado 1.07e-08	Invar	Acetic acid	secondary	3.50e-07	1.77e-08
Freming:Methanolsecondary3.36e.071.70e.08Dolyaymer Pord, high-impactNitarsecondary2.97e.071.55e.08Steel Prod, cold-rolled, semi-finishedVanndium (V3+, V5+)secondary2.95e.071.56e.08InvarCadmiumsecondary2.95e.071.46e.08InvarCadmiusecondary2.85e.071.46e.08InvarHepianesecondary2.76e.071.44e.08Steel Prod, cold-rolled, semi-finishedBoron (B HI)secondary2.76e.071.34e.08Steel Prod, cold-rolled, semi-finishedBoron (B HI)secondary2.26e.071.33e.08LadNay ProductionFacture (Inved isomery)secondary2.26e.071.33e.08LadNay ProductionFacture (Inved isomery)secondary2.12e.071.07e.08ABS ProductionHarongsecondary2.12e.071.07e.08ABS ProductionHarongsecondary2.12e.071.07e.08ABS ProductionLadsecondary2.12e.071.07e.08ABS ProductionLadsecondary2.12e.071.07e.08ABS ProductionLadsecondary2.12e.071.07e.08ABS ProductionLadsecondary2.12e.071.07e.08ABS ProductionNitrous oxidesecondary2.12e.071.07e.08ABS ProductionNitrous oxidesecondary2.12e.071.07e.08ABS ProductionNitrous oxidesecondary2.12e.071.07e.08St	Ferrite mfg.	Vanadium (V3+, V5+)	secondary	3.42e-07	1.73e-08
Polystyme Prod., high-impactNitratesecondary2.97e.071.53e.08Steel Prod., odd-villed, semi-finishedCadmium (V3+, V5+)secondary2.95e.071.50e.08Alumium Prod.Cadmiumsecondary2.85e.071.45e.08InvarHeptanesecondary2.76e.071.45e.08Ferrite mfg.Heptanesecondary2.76e.071.43e.08Steel Prod., cold-villed, semi-finishedBoron (B11)secondary2.56e.071.33e.08LadXyareActuidehydesecondary2.46e.071.30e.08LadXyareSecondary2.42e.071.23e.08LadXyareMarciny2.42e.071.23e.08InvarMercurysecondary2.12e.071.07e.08ABS ProductionFihanethiolsecondary2.12e.071.07e.08ABS ProductionHalogenatohydroarbons (unspecified)secondary2.12e.071.07e.08ABS ProductionHalogenatohydroarbons (unspecified)secondary2.12e.071.07e.08ABS ProductionMirous oxidesecondary2.12e.071.07e.08ABS ProductionMirous oxidesecondary2.12e.071.07e.08ABS ProductionMirous oxidesecondary2.12e.071.07e.08ABS ProductionMirous oxidesecondary2.12e.071.07e.08ABS ProductionMirous oxidesecondary2.12e.071.07e.08Styrene-butadiene Copolymer Prod.Halogens ulfrådesecondary2.07e.071.05e.0	Ferrite mfg.	Methanol	secondary	3.36e-07	1.70e-08
Steel Prod., cold-rolled, semi-finishedVanadium (V3+, V3+)secondary295-071.50-08InvarCyunide (-1)secondary2.88-071.46-08InvarHeptanesecondary2.78-071.40-08Forrite mfgHeptanesecondary2.78-071.40-08Steel Prod., cold-rolled, semi-finishedBoron (B III)secondary2.78-071.31-08Ads ProductionActaldehydesecondary2.24-071.32-08Ads ProductionHALON-1301secondary2.24-071.32-08Alaminur Prod.Trichtylen glycolsecondary2.12-071.07-08ABS ProductionHancarkiolsecondary2.12-071.07-08ABS ProductionHalogenatel hydrocarbons (usspecified)secondary2.12-071.07-08ABS ProductionLadosecondary2.12-071.07-08ABS ProductionLadosecondary2.12-071.07-08ABS ProductionLadosecondary2.12-071.07-08ABS ProductionLadosecondary2.12-071.07-08ABS ProductionKarcurysecondary2.12-071.07-08ABS ProductionKarcurysecondary2.12-071.07-08ABS ProductionKarcurysecondary2.12-071.07-08Syrene-butalized cophymer Prod.Halogenatel hydrocarbons (usspecified)secondary2.12-071.05-08Syrene-butalized cophymer Prod.Halogenatel hydrocarbons (usspecified)secondary2.12-071.05-08	Polystyrene Prod., high-impact	Nitrate	secondary	3.02e-07	1.53e-08
Aluminum Prod. Cadmium secondary 295-07 1.50-08 Invar Cyanide (-1) secondary 2.85-07 1.46-08 Invar Heptane secondary 2.76-07 1.37-08 Steel Prod., cold-stolled, seni-finished Boron (BII) secondary 2.56-07 1.37-08 ABS Production Acctaldely de secondary 2.56-07 1.39-08 ABS Production Natore (mixed isomers) secondary 2.54-07 1.29-08 Alumium Prod. Triethylene glycol secondary 2.12-07 1.08-08 ABS Production Fabarethiol secondary 2.12-07 1.07-08 ABS Production Halogenated hydrocarbons (unspecified) secondary 2.12-07 1.07-08 ABS Production Lead secondary 2.12-07 1.07-08 ABS Production Marcury secondary 2.12-07 1.07-08 ABS Production Marcury secondary 2.12-07 1.07-08 ABS Production Nitrous oxide secondary 2.12-07	Steel Prod., cold-rolled, semi-finished	Vanadium (V3+, V5+)	secondary	2.97e-07	1.50e-08
InvarCyanide (-1)secondary2.88-071.43-08InvarHeptanesecondary2.76e-071.43-08Steel Prod., cold-rolled, semi-finishedBoron (B III)secondary2.76e-071.34-08InvarActaldedhydsecondary2.56e-071.33-08AttS ProductionHALON-1301secondary2.56e-071.30-08AttS ProductionTrichlycne glycolsecondary2.54e-071.23e-08LeadXylene (mixed isomers)secondary2.12e-071.07e-08AttS ProductionHanchiolsecondary2.12e-071.07e-08AttS ProductionHalogenated phroearbos (unspecified)secondary2.12e-071.07e-08AttS ProductionLadsecondary2.12e-071.07e-08AttS ProductionLadosecondary2.12e-071.07e-08AttS ProductionLadosecondary2.12e-071.07e-08AttS ProductionKirous exidesecondary2.12e-071.07e-08AttS ProductionKirous exidesecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Halogenated phroearbos (unspecifie)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Leadsecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nitrous exidesecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nitrous exidesecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nitrous exi	Aluminum Prod.	Cadmium	secondary	2.95e-07	1.50e-08
InvarHeptanesecondary2.83e-071.43e-08Ferrite Infg.Heptanesecondary2.76e-071.37e-08InvarActaldelydesecondary2.64e-071.37e-08InvarActaldelydesecondary2.64e-071.33e-08ABS ProductionHLON-1301secondary2.54e-071.23e-08LeadXylenc (mixed isomers)secondary2.12e-071.07e-08Aluminum Prod.Finder Jegoolsecondary2.12e-071.07e-08ABS ProductionEbanethiolsecondary2.12e-071.07e-08ABS ProductionFinder Jegoolsecondary2.12e-071.07e-08ABS ProductionHalogenated hydrocarbons (unspecified)secondary2.12e-071.07e-08ABS ProductionLeadsecondary2.12e-071.07e-08ABS ProductionKercarysecondary2.12e-071.07e-08ABS ProductionKercarysecondary2.07e-071.05e-08Syrene-butadiene Copolyme Prod.Halogenated hydrocarbons (unspecified)secondary2.07e-071.05e-08Syrene-butadiene Copolymer Prod.Hydrogen sulfidesecondary2.07e-071.05e-08Syrene-butadiene Copolymer Prod.Kef (-2)secondary2.07e-071.05e-08Syrene-butadiene Copolymer Prod.Kef (-2)secondary2.07e-071.05e-08Syrene-butadiene Copolymer Prod.Kef (-2)secondary2.07e-071.05e-08Syrene-butadiene Copolymer Prod.Kef (-2)secondary<	Invar	Cyanide (-1)	secondary	2.89e-07	1.46e-08
Ferrite mfg.Heptanesecondary2.76e-071.40e-08Steel Prod., cold-rolled, semi-finishedBoron (B III)secondary2.264e-071.34e-08ARS ProductionHALON-1301secondary2.56e-071.30e-08Alls ProductionHALON-1301secondary2.54e-071.23e-08Aluminum Prod.Tricthylene glycolsecondary2.12e-071.07e-08ABS ProductionEthamethiolsecondary2.12e-071.07e-08ABS ProductionHalogenated hydrocarbons (unspecified)secondary2.12e-071.07e-08ABS ProductionLeadsecondary2.12e-071.07e-08ABS ProductionHalogenated hydrocarbons (unspecified)secondary2.12e-071.07e-08ABS ProductionMercurysecondary2.12e-071.07e-08ABS ProductionKinou soxidesecondary2.12e-071.07e-08ABS ProductionZine (†2)secondary2.12e-071.07e-08Styrene-butadiene Copolymer Prod.Halogenated hydrocarbons (unspecified)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Leadsecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Kicel (*2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nitrous soxidesecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Kicel (*2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nitrous soxidesecondary </td <td>Invar</td> <td>Heptane</td> <td>secondary</td> <td>2.83e-07</td> <td>1.43e-08</td>	Invar	Heptane	secondary	2.83e-07	1.43e-08
Steel Prod., cold-rolled, semi-finished Boron (B III) secondary 2.76e-07 1.37e-08 Invar Acetaldehyde secondary 2.64e-07 1.34e-08 ABS Production HALON-1301 secondary 2.54e-07 1.28e-08 Lead Xylene (mixed isomers) secondary 2.12e-07 1.08e-08 Invar Mercury secondary 2.12e-07 1.08e-08 ABS Production Ethamethiol secondary 2.12e-07 1.07e-08 ABS Production Halogenated hydrocarbons (unspecified) secondary 2.12e-07 1.07e-08 ABS Production Mercury secondary 2.12e-07 1.07e-08 ABS Production Mercury secondary 2.12e-07 1.07e-08 ABS Production Nitrous oxide secondary 2.07e-07 1.05e-08 Styrene-butadienc Copolymer Prod. Halogenated hydrocarbons (unspecified) secondary 2.07e-07 1.05e-08 Styrene-butadienc Copolymer Prod. Halogenated hydrocarbons (unspecified) secondary 2.07e-07 1.05e-08	Ferrite mfg.	Heptane	secondary	2.76e-07	1.40e-08
InvarAcetaldebydesecondary246-071.34-08ABS ProductionHALON-1301secondary2.56e-071.30-08LadXylene (mixed isomers)secondary2.42-071.23-08Aluminum Prod.Triethylene glycolsecondary2.12-071.07-08ABS ProductionHunefhiolsecondary2.12-071.07-08ABS ProductionHuoridesecondary2.12-071.07-08ABS ProductionLeadsecondary2.12-071.07-08ABS ProductionMercurysecondary2.12-071.07-08ABS ProductionMercurysecondary2.12-071.07-08ABS ProductionMercurysecondary2.12-071.07-08ABS ProductionNirous oxidesecondary2.12-071.07-08Styrene-butadiene Copolymer Prod.Halogenatel hydrocarbons (unspecifie)secondary2.07-071.05-08Styrene-butadiene Copolymer Prod.Leadsecondary2.07-071.05-08Styrene-butadiene Copolymer Prod.Leadsecondary2.07-071.05-08Styrene-butadiene Copolymer Prod.Nictel (2.2secondary2.07-071.05-08Styrene-butadiene Copolymer Prod.Nictel (2.2secondary2.07-071.05-08Styrene-butadiene Copolymer Prod.Nictel (2.2secondary2.07-071.05-08Styrene-butadiene Copolymer Prod.Nictel (2.2secondary2.07-071.05-08Styrene-butadiene Copolymer Prod.Nictel (2.2secondary <td>Steel Prod., cold-rolled, semi-finished</td> <td>Boron (B III)</td> <td>secondary</td> <td>2.70e-07</td> <td>1.37e-08</td>	Steel Prod., cold-rolled, semi-finished	Boron (B III)	secondary	2.70e-07	1.37e-08
ABS ProductionHALON-1301secondary2.56e-071.30e-08LeadXylene (mixed isomers)secondary2.54e-071.29-08Auminum Pod.Trictlylene glyc01secondary2.12e-071.08-08ABS ProductionEthanethiolsecondary2.12e-071.07-08ABS ProductionHalogenated hydrocarbons (unspecified)secondary2.12e-071.07-08ABS ProductionLeadsecondary2.12e-071.07-08ABS ProductionLeadsecondary2.12e-071.07-08ABS ProductionMercurysecondary2.12e-071.07-08ABS ProductionMercurysecondary2.12e-071.07-08ABS ProductionMercurysecondary2.12e-071.05-08Styrene-butadiene Copolymer Prod.Halogenated hydrocarbons (unspecified)secondary2.07e-071.05-08Styrene-butadiene Copolymer Prod.Halogenated hydrocarbons (unspecified)secondary2.07e-071.05-08Styrene-butadiene Copolymer Prod.Halogenated hydrocarbons (unspecified)secondary2.07e-071.05-08Styrene-butadiene Copolymer Prod.Halogenated hydrocarbons (unspecified)secondary2.07e-071.05-08Styrene-butadiene Copolymer Prod.Kicel (+2)secondary2.07e-071.05-08Styrene-butadiene Copolymer Prod.Kicel (+2)secondary2.07e-071.05-08Styrene-butadiene Copolymer Prod.Kicel (+2)secondary2.07e-071.05-08Styrene-butadiene Copolymer Prod.<	Invar	Acetaldehyde	secondary	2.64e-07	1.34e-08
LeadXylene (mixed isomers)secondary2.54e-071.29e-08Aluminum Prod.Triethylene glycolsecondary2.12e-071.08e-08InvarMercurysecondary2.12e-071.07e-08ABS ProductionEhanethiolsecondary2.12e-071.07e-08ABS ProductionHalogenated hydrocarbons (unspecified)secondary2.12e-071.07e-08ABS ProductionLeadsecondary2.12e-071.07e-08ABS ProductionMercurysecondary2.12e-071.07e-08ABS ProductionNitrous oxidesecondary2.12e-071.07e-08ABS ProductionXitrous oxidesecondary2.12e-071.07e-08ABS ProductionZinc (+2)secondary2.12e-071.07e-08Styrene-butadiene Copolymer Prod.Hydrogen suffidesecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Hydrogen suffidesecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08 </td <td>ABS Production</td> <td>HALON-1301</td> <td>secondary</td> <td>2.56e-07</td> <td>1.30e-08</td>	ABS Production	HALON-1301	secondary	2.56e-07	1.30e-08
Aluminum Prod.Tricthylene glycolsecondary2.42e-071.128-08InvarMercurysecondary2.12e-071.076-08ABS ProductionEhanchiolsecondary2.12e-071.070-08ABS ProductionHalogenated hydrocarbons (unspecified)secondary2.12e-071.070-08ABS ProductionLeadsecondary2.12e-071.070-08ABS ProductionLeadsecondary2.12e-071.070-08ABS ProductionMercurysecondary2.12e-071.070-08ABS ProductionNitrou socidesecondary2.12e-071.070-08ABS ProductionNitrou socidesecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Halogenated hydrocarbons (unspecified)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Halogenated hydrocarbons (unspecified)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Mercurysecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2) </td <td>Lead</td> <td>Xylene (mixed isomers)</td> <td>secondary</td> <td>2.54e-07</td> <td>1.29e-08</td>	Lead	Xylene (mixed isomers)	secondary	2.54e-07	1.29e-08
InvarMercurysecondary2.12-071.08-08ABS ProductionEthanethiolsecondary2.12-071.07-08ABS ProductionHalogenated hydrocarbons (unspecified)secondary2.12-071.07-08ABS ProductionLeadsecondary2.12-071.07-08ABS ProductionMercurysecondary2.12-071.07-08ABS ProductionMercurysecondary2.12-071.07-08ABS ProductionMercurysecondary2.12-071.07-08ABS ProductionZine (+2)secondary2.07-071.05-08Styrene-butadiene Copolymer Prod.Halogenated hydrocarbons (unspecified)secondary2.07-071.05-08Styrene-butadiene Copolymer Prod.Leadsecondary2.07-071.05-08Styrene-butadiene Copolymer Prod.Nikrou soxidesecondary2.07-071.05-08Styrene-butadiene Copolymer Prod.Nikrou soxidesecondary2.07-07	Aluminum Prod.	Triethylene glycol	secondary	2.42e-07	1.23e-08
ABS ProductionEithanethiolsecondary2.12e-071.07e-08ABS ProductionHalogenated hydrocarbons (unspecified)secondary2.12e-071.07e-08ABS ProductionLeadsecondary2.12e-071.07e-08ABS ProductionMercurysecondary2.12e-071.07e-08ABS ProductionMirous oxidesecondary2.12e-071.07e-08ABS ProductionXincus oxidesecondary2.12e-071.07e-08ABS ProductionZinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Halogenated hydrocarbons (unspecified)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Hadogenated hydrocarbons (unspecified)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Mercurysecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Xincu (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Xincu (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Xincu (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Xincu (+2)secondary2.07e-071.05e-08Steel Prod., cold-rolled, semi-finishedAcetia acidsecondary2.05e-071.04e-08Auminum Prod.Qanie (-1)secondary1.95e-079.06e-09InvarTriethylen	Invar	Mercury	secondary	2.12e-07	1.08e-08
ABS ProductionFluoridesecondary2.12e-071.07e-08ABS ProductionLeadsecondary2.12e-071.07e-08ABS ProductionLeadsecondary2.12e-071.07e-08ABS ProductionMercurysecondary2.12e-071.07e-08ABS ProductionNirrous oxidesecondary2.12e-071.07e-08ABS ProductionZinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Halogenated hydrocarbons (unspecified)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Leadsecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.05e-071.04e-08Aluminum Prod.Cyanide (-1)secondary2.05e-071.04e-08Aluminum Prod.Cyanide (-1)secondary1.95e-079.09e-09InvarTricthylene glycolsecondary1.95e-079.09e-09InvarTricthylene glycolsecondary1.95e-079.09e-09Invar <td< td=""><td>ABS Production</td><td>Ethanethiol</td><td>secondary</td><td>2.12e-07</td><td>1.07e-08</td></td<>	ABS Production	Ethanethiol	secondary	2.12e-07	1.07e-08
ABS ProductionHalogenated hydrocarbons (unspecified)secondary2.12e-071.07e-08ABS ProductionLeadsecondary2.12e-071.07e-08ABS ProductionMercurysecondary2.12e-071.07e-08ABS ProductionZinc (*2)secondary2.12e-071.07e-08ABS ProductionZinc (*2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Halogenated hydrocarbons (unspecified)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Leadsecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (*2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (*2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (*12)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (*12)secondary2.07e-071.04e-08Styrene-butadiene Copolymer Prod.Nickel (*12)secondary2.07e-071.04e-08Styrene-butadiene Copolymer Prod.Nickel (*1)secondary2.07e-071.04e-08Styrene-butadiene Copolymer Prod.Nickel (*1)secondary2.07e-071.04e-08Styrene-butadiene Copolymer Prod.Nickel (*1)secondary2.05e-071.04e-08Styrene-butadiene Copolymer Prod.Nickel (*1)secondary1.95e-079.06e-09InvarTriethylene glycolsecondary1.95e-079.06e-09InvarTrieth	ABS Production	Fluoride	secondary	2.12e-07	1.07e-08
ABS ProductionLeadsecondary2.12e-071.07e-08ABS ProductionMercurysecondary2.12e-071.07e-08ABS ProductionXincou sxidesecondary2.12e-071.07e-08ABS ProductionZinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Halogenated hydrocarbons (unspecified)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Leadsecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Leadsecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+1)secondary2.05e-071.04e-08Stel Prod., cold-rolled, semi-finishedMercurysecondary1.95e-079.90e-09InvarTriethylene g	ABS Production	Halogenated hydrocarbons (unspecified)	secondary	2.12e-07	1.07e-08
ABS ProductionMercurysecondary2.12e-071.07e-08ABS ProductionNitrous oxidesecondary2.12e-071.07e-08ABS ProductionZinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Halogenated hydrocarbons (unspecified)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Leadsecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Leadsecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+1)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.04e-08Steel Prod, cold-rolled, semi-finishedAcetic acidsecondary1.05e-079.09e-09InvarTriethylene glycolsecondary1.92e-079.09e-09InvarTriethylene glycolsecondary1.92e-079.09e-09InvarTriethylene glycolsecondary1.92e-079.09e-09InvarTriethylene glycolsecondary <td>ABS Production</td> <td>Lead</td> <td>secondary</td> <td>2.12e-07</td> <td>1.07e-08</td>	ABS Production	Lead	secondary	2.12e-07	1.07e-08
ABS ProductionNitrous oxidesecondary2.12e-071.07e-08ABS ProductionZinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Halogenatel hydrocarbons (unspecified)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Leadsecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Mercurysecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+1)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.05e-071.04e-08Steel Prod., cold-rolled, semi-finishedAcetic acidsecondary1.95e-079.97e-09InvarTriethylene glycolsecondary1.95e-079.97e-09Steel Prod., cold-rolled, semi-finishedTriethylene glycolsecondary1.89e-079.97e-09Aluminum Prod.Cyanide (-1)secondary1.89e-079.97e-09Steel Prod., cold-rolled, semi-finishedTriethylene glycolsecondary1.89e-079.	ABS Production	Mercury	secondary	2.12e-07	1.07e-08
ABS ProductionZinc (+2)secondary2.12e-071.07e-08Styrene-butadiene Copolymer Prod.Halgenated hydrocarbons (unspecified)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Leadsecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Leadsecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.04e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.04e-08Styrene-butadiene Copolymer Prod.Zinc (-1)secondary2.07e-071.04e-08Aluminum Prod.Mercurysecondary1.92e-079.97e-09InvarTriethylene glycolsecondary1.92e-079.97e-09InvarTriethylene glycolsecondary1.88e-079.48e-09Aluminum Prod.Cyanide (-1)secondary1.88e-079.48e-09LeadTriethylene glycolsecondary1.88e-079.48e-09LeadTriethylene glycolsecondary1.88e-0	ABS Production	Nitrous oxide	secondary	2.12e-07	1.07e-08
Styrene-butadiene Copolymer Prod.Halogenated hydrocarbons (unspecified)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Leadsecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Mercurysecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Mirous oxidesecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nirous oxidesecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nirous oxidesecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.05e-071.04e-08Styrene-butadiene Copolymer Prod.Zinc (+1)secondary2.05e-071.04e-08Styrene-butadiene Copolymer Prod.Mercurysecondary2.05e-071.04e-08Steel Prod., cold-rolled, semi-finishedAcetic acidsecondary2.05e-079.09e-09InvarTriethylene glycolsecondary1.95e-079.09e-09InvarTriethylene glycolsecondary1.95e-079.09e-09Steel Prod., cold-rolled, semi-finishedTriethylene glycolsecondary1.95e-079.06e-09LeadTriethylene glycolsecondary1.88e-079.16e-09Aluminum Prod.HALON-1301secondary1.88e-079.16e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.77e-078.98e-09 <td>ABS Production</td> <td>Zinc (+2)</td> <td>secondary</td> <td>2.12e-07</td> <td>1.07e-08</td>	ABS Production	Zinc (+2)	secondary	2.12e-07	1.07e-08
Styrene-butadiene Copolymer Prod.Hydrogen sulfidesecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Leadsecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.05e-071.04e-08Steel Prod., cold-rolled, semi-finishedAcetic acidsecondary2.05e-071.04e-08Aluminum Prod.Mercurysecondary2.05e-079.00e-09InvarTriethylene glycolsecondary1.95e-079.00e-09InvarTriethylene glycolsecondary1.92e-079.70e-09Aluminum Prod.Cyanide (-1)secondary1.92e-079.70e-09LeadTriethylene glycolsecondary1.88e-079.54e-09LeadTriethylene glycolsecondary1.88e-079.54e-09LeadTriethylene glycolsecondary1.88e-079.14e-09Aluminum Prod.HALON-1301secondary1.79e-078.98e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.79e-078.98e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.78e-078.98e	Styrene-butadiene Copolymer Prod.	Halogenated hydrocarbons (unspecified)	secondary	2.07e-07	1.05e-08
Styrene-butadiene Copolymer Prod.Leadsecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zine (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zine (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zine (+2)secondary2.05e-071.04e-08Styrene-butadiene Copolymer Prod.Mercurysecondary2.05e-071.04e-08Steel Prod., cold-rolled, semi-finishedAcetic acidsecondary2.05e-071.02e-08Ferrite mfg.Phenolsecondary1.93e-079.00e-09InvarTriethylene glycolsecondary1.92e-079.77e-09Steel Prod., cold-rolled, semi-finishedTriethylene glycolsecondary1.92e-079.06e-09InvarTriethylene glycolsecondary1.88e-079.40e-09LeadTriethylene glycolsecondary1.88e-079.14e-09Aluminum Prod.HALON-1301secondary1.80e-079.14e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-fin	Styrene-butadiene Copolymer Prod.	Hydrogen sulfide	secondary	2.07e-07	1.05e-08
Styrene-butadiene Copolymer Prod.Mercurysecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.05e-071.04e-08Steel Prod., cold-rolled, semi-finishedAcetic acidsecondary2.05e-071.04e-08Aluminum Prod.Mercurysecondary2.02e-071.02e-08Ferrite mfg.Phenolsecondary1.95e-079.00e-09InvarTriethylene glycolsecondary1.92e-079.70e-09Steel Prod., cold-rolled, semi-finishedTriethylene glycolsecondary1.92e-079.70e-09Aluminum Prod.Cyanide (-1)secondary1.89e-079.60e-09Ferrite mfg.Triethylene glycolsecondary1.88e-079.54e-09LeadTriethylene glycolsecondary1.88e-079.18e-09Aluminum Prod.HALON-1301secondary1.80e-079.18e-09Steel Prod., sigh-impactPhenolsecondary1.79e-078.98e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.79e-078.98e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedKontiumsecondary	Styrene-butadiene Copolymer Prod.	Lead	secondary	2.07e-07	1.05e-08
Styrene-butadiene Copolymer Prod.Nickel (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.04e-08Ferrite mfg.Cyanide (-1)secondary2.05e-071.04e-08Steel Prod., cold-rolled, semi-finishedAcetic acidsecondary2.02e-071.02e-08Ferrite mfg.Phenolsecondary2.02e-079.90e-09InvarTriethylene glycolsecondary1.95e-079.90e-09InvarTriethylene glycolsecondary1.92e-079.70e-09Steel Prod., cold-rolled, semi-finishedTriethylene glycolsecondary1.89e-079.60e-09Ferrite mfg.Triethylene glycolsecondary1.88e-079.54e-09LeadTriethylene glycolsecondary1.81e-079.14e-09Aluminum Prod.HALON-1301secondary1.80e-079.14e-09Polystyrene Prod., high-impactPhenolsecondary1.79e-078.98e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.78e-078.98e-09Steel Prod., cold-rolled, semi-finishedCadmiums	Styrene-butadiene Copolymer Prod.	Mercury	secondary	2.07e-07	1.05e-08
Styrene-butadiene Copolymer Prod.Nitrous oxidesecondary2.07e-071.05e-08Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.05e-08Ferrite mfg.Cyanide (-1)secondary2.05e-071.04e-08Steel Prod., cold-rolled, semi-finishedAcetic acidsecondary2.02e-071.02e-08Aluminum Prod.Mercurysecondary2.02e-071.02e-08Ferrite mfg.Phenolsecondary1.93e-079.70e-09InvarTriethylene glycolsecondary1.92e-079.70e-09Aluminum Prod.Cyanide (-1)secondary1.89e-079.60e-09Ferrite mfg.Triethylene glycolsecondary1.89e-079.54e-09LeadTriethylene glycolsecondary1.80e-079.18e-09Aluminum Prod.HALON-1301secondary1.80e-079.14e-09Polystyrene Prod., high-impactPhenolsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedPropylenesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedKorontiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedKorontiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedKorontiumsecondary <td< td=""><td>Styrene-butadiene Copolymer Prod.</td><td>Nickel (+2)</td><td>secondary</td><td>2.07e-07</td><td>1.05e-08</td></td<>	Styrene-butadiene Copolymer Prod.	Nickel (+2)	secondary	2.07e-07	1.05e-08
Styrene-butadiene Copolymer Prod.Zinc (+2)secondary2.07e-071.05e-08Ferrite mfg.Cyanide (-1)secondary2.05e-071.04e-08Steel Prod., cold-rolled, semi-finishedAcetic acidsecondary2.02e-071.02e-08Aluminum Prod.Mercurysecondary2.02e-071.02e-08Ferrite mfg.Phenolsecondary1.95e-079.09e-09InvarTriethylene glycolsecondary1.92e-079.77e-09Steel Prod., cold-rolled, semi-finishedTriethylene glycolsecondary1.89e-079.60e-09Ferrite mfg.Triethylene glycolsecondary1.89e-079.60e-09Inuminum Prod.Cyanide (-1)secondary1.88e-079.54e-09LeadTriethylene glycolsecondary1.80e-079.14e-09Polystyrene Prod., high-impactPhenolsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedPropylenesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedKorntiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedKorntiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedKorntiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedKor	Styrene-butadiene Copolymer Prod.	Nitrous oxide	secondary	2.07e-07	1.05e-08
Ferrite mfg.Cyanide (-1)secondary2.05e-071.04e-08Steel Prod., cold-rolled, semi-finishedAcetic acidsecondary2.05e-071.04e-08Aluminum Prod.Mercurysecondary2.02e-071.02e-08Ferrite mfg.Phenolsecondary1.95e-079.90e-09InvarTriethylene glycolsecondary1.93e-079.77e-09Steel Prod., cold-rolled, semi-finishedTriethylene glycolsecondary1.92e-079.70e-09Aluminum Prod.Cyanide (-1)secondary1.89e-079.60e-09Ferrite mfg.Triethylene glycolsecondary1.89e-079.50e-09LeadTriethylene glycolsecondary1.80e-079.14e-09Aluminum Prod.HALON-1301secondary1.80e-079.14e-09Polystyrene Prod., high-impactPhenolsecondary1.79e-079.08e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.76e-078.18e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.76e-078.18e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.76e-078.28e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary </td <td>Styrene-butadiene Copolymer Prod.</td> <td>Zinc (+2)</td> <td>secondary</td> <td>2.07e-07</td> <td>1.05e-08</td>	Styrene-butadiene Copolymer Prod.	Zinc (+2)	secondary	2.07e-07	1.05e-08
Steel Prod., cold-rolled, semi-finishedAcetic acidsecondary2.05e-071.04e-08Aluminum Prod.Mercurysecondary2.02e-071.02e-08Ferrite mfg.Phenolsecondary1.95e-079.90e-09InvarTriethylene glycolsecondary1.92e-079.77e-09Steel Prod., cold-rolled, semi-finishedTriethylene glycolsecondary1.92e-079.70e-09Aluminum Prod.Cyanide (-1)secondary1.89e-079.60e-09Ferrite mfg.Triethylene glycolsecondary1.88e-079.54e-09LeadTriethylene glycolsecondary1.80e-079.18e-09Aluminum Prod.HALON-1301secondary1.80e-079.14e-09Polystyrene Prod., high-impactPhenolsecondary1.80e-079.11e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedProylenesecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.62e-078.23e-09LeadMetnanolsecondary1.62e-078.23e-09Ferrite mfg.Mercurysecondary1.62e-078.18e-09Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary1.52e-077.70e-09<	Ferrite mfg.	Cyanide (-1)	secondary	2.05e-07	1.04e-08
Aluminum Prod.Mercurysecondary2.02e-071.02e-08Ferrite mfg.Phenolsecondary1.95e-079.90e-09InvarTriethylene glycolsecondary1.93e-079.77e-09Steel Prod., cold-rolled, semi-finishedTriethylene glycolsecondary1.92e-079.70e-09Aluminum Prod.Cyanide (-1)secondary1.89e-079.60e-09Ferrite mfg.Triethylene glycolsecondary1.88e-079.54e-09LeadTriethylene glycolsecondary1.81e-079.18e-09Aluminum Prod.HALON-1301secondary1.80e-079.14e-09Polystyrene Prod., high-impactPhenolsecondary1.79e-079.08e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedPropylenesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedAtminumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedAtminumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedAtminumsecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary1.52e-077.76e-09LeadMethanolsecondary1.52e-077.76e-09Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary<	Steel Prod., cold-rolled, semi-finished	Acetic acid	secondary	2.05e-07	1.04e-08
Ferrite mfg.Phenolsecondary1.95e-079.90e-09InvarTriethylene glycolsecondary1.93e-079.77e-09Steel Prod., cold-rolled, semi-finishedTriethylene glycolsecondary1.92e-079.70e-09Aluminum Prod.Cyanide (-1)secondary1.89e-079.60e-09Ferrite mfg.Triethylene glycolsecondary1.88e-079.54e-09LeadTriethylene glycolsecondary1.81e-079.18e-09Aluminum Prod.HALON-1301secondary1.80e-079.14e-09Polystyrene Prod., high-impactPhenolsecondary1.79e-079.08e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedPropylenesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.72e-078.98e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.72e-078.98e-09LeadMetnanolsecondary1.72e-078.98e-09Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary1.72e-078.98e-09Steel Prod., cold-rolled, semi-finishedAcetaldehyd	Aluminum Prod.	Mercury	secondary	2.02e-07	1.02e-08
InvarTriethylene glycolsecondary1.93e-079.77e-09Steel Prod., cold-rolled, semi-finishedTriethylene glycolsecondary1.92e-079.70e-09Aluminum Prod.Cyanide (-1)secondary1.89e-079.60e-09Ferrite mfg.Triethylene glycolsecondary1.88e-079.54e-09LeadTriethylene glycolsecondary1.81e-079.18e-09Aluminum Prod.HALON-1301secondary1.80e-079.14e-09Polystyrene Prod., high-impactPhenolsecondary1.79e-079.08e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.72e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.72e-078.93e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.72e-078.93e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.72e-078.32e-09Ferrite mfg.Mercurysecondary1.62e-078.18e-09Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary1.52e-077.76e-09ABS ProductionEthylenesecondary1.52e-077.70e-09LeadPhenolsecondary1.37e-076.96e-09InvarEthylbenzenesecondary1.37e-076.92e-09 <td>Ferrite mfg.</td> <td>Phenol</td> <td>secondary</td> <td>1.95e-07</td> <td>9.90e-09</td>	Ferrite mfg.	Phenol	secondary	1.95e-07	9.90e-09
Steel Prod., cold-rolled, semi-finishedTriethylene glycolsecondary1.92e-079.70e-09Aluminum Prod.Cyanide (-1)secondary1.89e-079.60e-09Ferrite mfg.Triethylene glycolsecondary1.88e-079.54e-09LeadTriethylene glycolsecondary1.81e-079.18e-09Aluminum Prod.HALON-1301secondary1.80e-079.14e-09Polystyrene Prod., high-impactPhenolsecondary1.80e-079.11e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedPropylenesecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.74e-078.83e-09LeadMethanolsecondary1.62e-078.18e-09Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary1.53e-077.76e-09ABS ProductionEthylenesecondary1.53e-077.70e-09LeadPhenolsecondary1.37e-076.96e-09InvarEthylenzenesecondary1.37e-076.92e-09	Invar	Triethylene glycol	secondary	1.93e-07	9.77e-09
Aluminum Prod.Cyanide (-1)secondary1.89e-079.60e-09Ferrite mfg.Triethylene glycolsecondary1.88e-079.54e-09LeadTriethylene glycolsecondary1.81e-079.18e-09Aluminum Prod.HALON-1301secondary1.80e-079.14e-09Polystyrene Prod., high-impactPhenolsecondary1.80e-079.11e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.79e-079.08e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedPropylenesecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.62e-078.83e-09LeadMethanolsecondary1.62e-078.18e-09Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary1.53e-077.76e-09ABS ProductionEthylenesecondary1.52e-077.70e-09LeadPhenolsecondary1.52e-077.70e-09LeadPhenolsecondary1.52e-077.70e-09ABS ProductionEthylenzenesecondary1.37e-076.96e-09InvarEthylbenzenesecondary1.37e-076.92e-09	Steel Prod., cold-rolled, semi-finished	Triethylene glycol	secondary	1.92e-07	9.70e-09
Ferrite mfg.Triethylene glycolsecondary1.88e-079.54e-09LeadTriethylene glycolsecondary1.81e-079.18e-09Aluminum Prod.HALON-1301secondary1.80e-079.14e-09Polystyrene Prod., high-impactPhenolsecondary1.80e-079.11e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.79e-079.08e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedPropylenesecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.74e-078.83e-09LeadMethanolsecondary1.62e-078.23e-09Ferrite mfg.Mercurysecondary1.52e-077.76e-09ABS ProductionEthylenesecondary1.52e-077.70e-09LeadPhenolsecondary1.52e-077.70e-09ABS ProductionEthylenesecondary1.37e-076.92e-09InvarEthylenzenesecondary1.37e-076.92e-09	Aluminum Prod.	Cyanide (-1)	secondary	1.89e-07	9.60e-09
LeadTriethylene glycolsecondary1.81e-079.18e-09Aluminum Prod.HALON-1301secondary1.80e-079.14e-09Polystyrene Prod., high-impactPhenolsecondary1.80e-079.11e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.79e-079.08e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedPropylenesecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.74e-078.83e-09LeadMethanolsecondary1.62e-078.23e-09Ferrite mfg.Mercurysecondary1.62e-078.18e-09Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary1.53e-077.76e-09ABS ProductionEthylenesecondary1.52e-077.70e-09LeadPhenolsecondary1.37e-076.92e-09InvarEthylbenzenesecondary1.37e-076.92e-09	Ferrite mfg.	Triethylene glycol	secondary	1.88e-07	9.54e-09
Aluminum Prod.HALON-1301secondary1.80e-079.14e-09Polystyrene Prod., high-impactPhenolsecondary1.80e-079.11e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.79e-079.08e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedPropylenesecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.74e-078.83e-09LeadMethanolsecondary1.62e-078.23e-09Ferrite mfg.Mercurysecondary1.62e-078.18e-09Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary1.53e-077.76e-09Berol, cold-rolled, semi-finishedPropylenesecondary1.52e-077.70e-09LeadPhenolsecondary1.52e-077.70e-09ABS ProductionEthylenesecondary1.37e-076.96e-09InvarEthylbenzenesecondary1.37e-076.92e-09	Lead	Triethylene glycol	secondary	1.81e-07	9.18e-09
Polystyrene Prod., high-impactPhenolsecondary1.80e-079.11e-09Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.79e-079.08e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedPropylenesecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.74e-078.83e-09LeadMethanolsecondary1.62e-078.23e-09Ferrite mfg.Mercurysecondary1.62e-078.18e-09Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary1.53e-077.76e-09ABS ProductionEthylenesecondary1.52e-077.70e-09LeadPhenolsecondary1.37e-076.96e-09InvarEthylbenzenesecondary1.37e-076.92e-09	Aluminum Prod.	HALON-1301	secondary	1.80e-07	9.14e-09
Steel Prod., cold-rolled, semi-finishedMorpholinesecondary1.79e-079.08e-09Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedPropylenesecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.74e-078.83e-09LeadMethanolsecondary1.62e-078.23e-09Ferrite mfg.Mercurysecondary1.62e-078.18e-09Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary1.53e-077.76e-09ABS ProductionEthylenesecondary1.52e-077.70e-09LeadPhenolsecondary1.37e-076.96e-09InvarEthylbenzenesecondary1.37e-076.92e-09	Polystyrene Prod., high-impact	Phenol	secondary	1.80e-07	9.11e-09
Steel Prod., cold-rolled, semi-finishedStrontiumsecondary1.77e-078.98e-09Steel Prod., cold-rolled, semi-finishedPropylenesecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.74e-078.83e-09LeadMethanolsecondary1.62e-078.23e-09Ferrite mfg.Mercurysecondary1.62e-078.18e-09Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary1.53e-077.76e-09Steel Prod., cold-rolled, semi-finishedEthylenesecondary1.52e-077.70e-09LeadPhenolsecondary1.37e-076.96e-09InvarEthylbenzenesecondary1.37e-076.92e-09	Steel Prod., cold-rolled, semi-finished	Morpholine	secondary	1.79e-07	9.08e-09
Steel Prod., cold-rolled, semi-finishedPropylenesecondary1.77e-078.97e-09Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.74e-078.83e-09LeadMethanolsecondary1.62e-078.23e-09Ferrite mfg.Mercurysecondary1.62e-078.18e-09Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary1.53e-077.76e-09ABS ProductionEthylenesecondary1.52e-077.70e-09LeadPhenolsecondary1.37e-076.96e-09InvarEthylbenzenesecondary1.37e-076.92e-09	Steel Prod., cold-rolled, semi-finished	Strontium	secondary	1.77e-07	8.98e-09
Steel Prod., cold-rolled, semi-finishedCadmiumsecondary1.74e-078.83e-09LeadMethanolsecondary1.62e-078.23e-09Ferrite mfg.Mercurysecondary1.62e-078.18e-09Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary1.53e-077.76e-09ABS ProductionEthylenesecondary1.52e-077.70e-09LeadPhenolsecondary1.37e-076.96e-09InvarEthylbenzenesecondary1.37e-076.92e-09	Steel Prod., cold-rolled, semi-finished	Propylene	secondary	1.77e-07	8.97e-09
LeadMethanolsecondary1.62e-078.23e-09Ferrite mfg.Mercurysecondary1.62e-078.18e-09Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary1.53e-077.76e-09ABS ProductionEthylenesecondary1.52e-077.70e-09LeadPhenolsecondary1.37e-076.96e-09InvarEthylbenzenesecondary1.37e-076.92e-09	Steel Prod., cold-rolled, semi-finished	Cadmium	secondary	1.74e-07	8.83e-09
Ferrite mfg.Mercurysecondary1.62e-078.18e-09Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary1.53e-077.76e-09ABS ProductionEthylenesecondary1.52e-077.70e-09LeadPhenolsecondary1.37e-076.96e-09InvarEthylbenzenesecondary1.37e-076.92e-09	Lead	Methanol	secondary	1.62e-07	8.23e-09
Steel Prod., cold-rolled, semi-finishedAcetaldehydesecondary1.53e-077.76e-09ABS ProductionEthylenesecondary1.52e-077.70e-09LeadPhenolsecondary1.37e-076.96e-09InvarEthylbenzenesecondary1.37e-076.92e-09	Ferrite mfg.	Mercury	secondary	1.62e-07	8.18e-09
ABS ProductionEthylenesecondary1.52e-077.70e-09LeadPhenolsecondary1.37e-076.96e-09InvarEthylbenzenesecondary1.37e-076.92e-09	Steel Prod., cold-rolled, semi-finished	Acetaldehyde	secondary	1.53e-07	7.76e-09
LeadPhenolsecondary1.37e-076.96e-09InvarEthylbenzenesecondary1.37e-076.92e-09	ABS Production	Ethylene	secondary	1.52e-07	7.70e-09
Invar Ethylbenzene secondary 1.37e-07 6.92e-09	Lead	Phenol	secondary	1.37e-07	6.96e-09
	Invar	Ethylbenzene	secondary	1.37e-07	6.92e-09

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Invar	Acetone	secondary	1.36e-07	6.90e-09
Aluminum Prod.	Chromium (III)	secondary	1.31e-07	6.64e-09
Steel Prod., cold-rolled, semi-finished	Rubidium ion (Rb+)	secondary	1.21e-07	6.13e-09
Ferrite mfg.	Cobalt	secondary	1.20e-07	6.09e-09
Ferrite mfg.	Manganese	secondary	1.19e-07	6.01e-09
Steel Prod., cold-rolled, semi-finished	Fluorine	secondary	1.14e-07	5.76e-09
Lead	Cyanide (-1)	secondary	9.76e-08	4.94e-09
Lead	Ethylene	secondary	9.43e-08	4.78e-09
Invar	Strontium	secondary	9.37e-08	4.75e-09
	Molybdenum (Mo II, Mo III, Mo IV, Mo V,			
Steel Prod., cold-rolled, semi-finished	Mo VI)	secondary	9.07e-08	4.59e-09
Steel Prod., cold-rolled, semi-finished	Benzo[a]pyrene	secondary	8.92e-08	4.52e-09
Ferrite mfg.	Boron (B III)	secondary	8.91e-08	4.51e-09
Ferrite mfg.	Cobalt (Co I, Co II, Co III)	secondary	8.79e-08	4.45e-09
Ferrite mfg.	Acetic acid	secondary	8.74e-08	4.43e-09
Lead	Toluene	secondary	8.24e-08	4.18e-09
Steel Prod., cold-rolled, semi-finished	Acetone	secondary	7.91e-08	4.01e-09
Invar	HALON-1301	secondary	7.60e-08	3.85e-09
Lead	Cobalt	secondary	7.32e-08	3.71e-09
Lead	Acetic acid	secondary	7.30e-08	3.70e-09
Lead	Aromatic hydrocarbons	secondary	6.79e-08	3.44e-09
Invar	Sulfuric acid	secondary	5.98e-08	3.03e-09
Ferrite mfg.	Sulfuric acid	secondary	5.84e-08	2.96e-09
Invar	Mercury compounds	secondary	5.57e-08	2.82e-09
Ferrite mfg.	Mercury compounds	secondary	5.44e-08	2.76e-09
Steel Prod., cold-rolled, semi-finished	Lanthanum	secondary	5.37e-08	2.72e-09
Steel Prod., cold-rolled, semi-finished	Ethylbenzene	secondary	5.25e-08	2.66e-09
Invar	Morpholine	secondary	5.07e-08	2.57e-09
Invar	Propylene	secondary	5.03e-08	2.55e-09
Ferrite mfg.	Morpholine	secondary	4.95e-08	2.51e-09
Ferrite mfg.	Strontium	secondary	4.94e-08	2.50e-09
Ferrite mfg.	Propylene	secondary	4.91e-08	2.49e-09
Invar	Nitrous oxide	secondary	4.85e-08	2.46e-09
Invar	Chromium (III)	secondary	4.79e-08	2.42e-09
Lead	Ethylbenzene	secondary	4.72e-08	2.39e-09
Steel Prod., cold-rolled, semi-finished	Beryllium	secondary	4.72e-08	2.39e-09
Aluminum Prod.	Strontium	secondary	4.48e-08	2.27e-09
Ferrite mfg.	Acetaldehyde	secondary	4.38e-08	2.22e-09
Lead	Chromium (III)	secondary	3.86e-08	1.96e-09
Invar	Rubidium ion (Rb+)	secondary	3.66e-08	1.85e-09
Ferrite mfg.	Rubidium ion (Rb+)	secondary	3.57e-08	1.81e-09
Lead	Acetaldehyde	secondary	3.39e-08	1.72e-09
Lead	HALON-1301	secondary	3.39e-08	1.72e-09
Steel Prod., cold-rolled, semi-finished	Mercury	secondary	3.26e-08	1.65e-09
Invar	Fluorine	secondary	3.22e-08	1.63e-09
Ferrite mfg.	Fluorine	secondary	3.14e-08	1.59e-09
Steel Prod., cold-rolled, semi-finished	Thorium	secondary	2.88e-08	1.46e-09
	Molybdenum (Mo II, Mo III, Mo IV, Mo V,			
Ferrite mfg.	Mo VI)	secondary	2.55e-08	1.29e-09
Steel Prod., cold-rolled, semi-finished	HALON-1301	secondary	2.52e-08	1.28e-09

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Ferrite mfg.	Benzo[a]pyrene	secondary	2.48e-08	1.25e-09
Ferrite mfg.	Acetone	secondary	2.23e-08	1.13e-09
Ferrite mfg.	Ethylbenzene	secondary	1.94e-08	9.82e-10
Lead	Acetone	secondary	1.72e-08	8.70e-10
Steel Prod., cold-rolled, semi-finished	Tin	secondary	1.68e-08	8.49e-10
Steel Prod., cold-rolled, semi-finished	Phosphorus pentoxide	secondary	1.61e-08	8.15e-10
Invar	Lanthanum	secondary	1.53e-08	7.76e-10
Ferrite mfg.	Lanthanum	secondary	1.50e-08	7.58e-10
Invar	Beryllium	secondary	1.34e-08	6.80e-10
Ferrite mfg.	Beryllium	secondary	1.31e-08	6.64e-10
Steel Prod., cold-rolled, semi-finished	Scandium	secondary	1.31e-08	6.62e-10
Steel Prod., cold-rolled, semi-finished	Chlorine	secondary	1.14e-08	5.78e-10
Aluminum Prod.	1,2-Dichlorotetrafluoroethane	secondary	1.04e-08	5.28e-10
Ferrite mfg.	HALON-1301	secondary	9.52e-09	4.82e-10
Steel Prod., cold-rolled, semi-finished	Thallium	secondary	8.89e-09	4.51e-10
Aluminum Prod.	Perfluoroethane	secondary	8.46e-09	4.28e-10
Invar	Thorium	secondary	8.23e-09	4.17e-10
Ferrite mfg.	Thorium	secondary	8.03e-09	4.07e-10
Invar	Hydrogen cyanide	secondary	7.58e-09	3.84e-10
Ferrite mfg.	Hydrogen cyanide	secondary	7.40e-09	3.75e-10
Steel Prod., cold-rolled, semi-finished	Chromium (III)	secondary	5.29e-09	2.68e-10
Steel Prod., cold-rolled, semi-finished	Nitrites	secondary	5.19e-09	2.63e-10
Ferrite mfg.	Chromium (III)	secondary	5.01e-09	2.54e-10
Invar	Tin	secondary	4.77e-09	2.41e-10
Invar	Phosphorus pentoxide	secondary	4.69e-09	2.38e-10
Ferrite mfg.	Tin	secondary	4.65e-09	2.36e-10
Ferrite mfg.	Phosphorus pentoxide	secondary	4.58e-09	2.32e-10
Invar	1,2-Dichlorotetrafluoroethane	secondary	4.31e-09	2.19e-10
Invar	Scandium	secondary	3.74e-09	1.89e-10
Ferrite mfg.	Scandium	secondary	3.65e-09	1.85e-10
Invar	Perfluoromethane	secondary	2.85e-09	1.45e-10
Ferrite mfg.	Perfluoromethane	secondary	2.78e-09	1.41e-10
Invar	Thallium	secondary	2.55e-09	1.29e-10
Ferrite mfg.	Thallium	secondary	2.49e-09	1.26e-10
Invar	Trichlorofluoromethane	secondary	2.06e-09	1.04e-10
Lead	Benzo[a]pyrene	secondary	1.96e-09	9.94e-11
Steel Prod., cold-rolled, semi-finished	Edetic acid (EDTA)	secondary	1.93e-09	9.78e-11
Invar	Hypochlorous acid	secondary	1.77e-09	8.98e-11
Steel Prod., cold-rolled, semi-finished	Hypochlorous acid	secondary	1.76e-09	8.91e-11
Ferrite mfg.	Hypochlorous acid	secondary	1.73e-09	8.76e-11
Steel Prod., cold-rolled, semi-finished	Cobalt (Co I, Co II, Co III)	secondary	1.57e-09	7.96e-11
Invar	Dichloromethane	secondary	1.46e-09	7.39e-11
Steel Prod., cold-rolled, semi-finished	Dichloromethane	secondary	1.45e-09	7.34e-11
Ferrite mfg.	Dichloromethane	secondary	1.42e-09	7.21e-11
Invar	Chlorine	secondary	1.11e-09	5.62e-11
Ferrite mfg.	Chlorine	secondary	1.08e-09	5.49e-11
Ferrite mfg.	Trichlorofluoromethane	secondary	7.65e-10	3.87e-11
Invar	Dichlorodifluoromethane	secondary	6.37e-10	3.23e-11
Invar	Edetic acid (EDTA)	secondary	5.47e-10	2.77e-11
Ferrite mfg.	Edetic acid (EDTA)	secondary	5.34e-10	2.70e-11

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Invar	CFC-13	secondary	5.04e-10	2.55e-11
Steel Prod., cold-rolled, semi-finished	Isopropylpropionate	secondary	2.02e-10	1.02e-11
Ferrite mfg.	Isopropylpropionate	secondary	1.99e-10	1.01e-11
Invar	Halogenated matter (organic)	secondary	1.37e-10	6.95e-12
Ferrite mfg.	Halogenated matter (organic)	secondary	1.34e-10	6.78e-12
Steel Prod., cold-rolled, semi-finished	Lithium salts	secondary	9.91e-11	5.02e-12
Invar	Ethanethiol	secondary	8.71e-11	4.41e-12
Ferrite mfg.	Ethanethiol	secondary	8.50e-11	4.31e-12
Ferrite mfg.	Tin (Sn++, Sn4+)	secondary	7.29e-11	3.69e-12
Invar	Chloroform	secondary	5.44e-11	2.76e-12
Steel Prod., cold-rolled, semi-finished	Chloroform	secondary	5.40e-11	2.73e-12
Ferrite mfg.	Chloroform	secondary	5.31e-11	2.69e-12
Steel Prod., cold-rolled, semi-finished	Zirconium	secondary	4.66e-11	2.36e-12
Steel Prod., cold-rolled, semi-finished	Perfluoromethane	secondary	4.49e-11	2.28e-12
ABS Production	HCFC-22	secondary	3.47e-11	1.76e-12
Invar	Lithium salts	secondary	2.81e-11	1.42e-12
Ferrite mfg.	Lithium salts	secondary	2.74e-11	1.39e-12
Lead	Acrolein	secondary	2.38e-11	1.21e-12
Invar	Zirconium	secondary	1.34e-11	6.77e-13
Ferrite mfg.	Zirconium	secondary	1.30e-11	6.61e-13
Lead	Benzaldehyde	secondary	1.04e-12	5.25e-14
Invar	Trichloroethylene	secondary	7.82e-13	3.96e-14
Steel Prod., cold-rolled, semi-finished	Trichloroethylene	secondary	7.75e-13	3.93e-14
Ferrite mfg.	Trichloroethylene	secondary	7.63e-13	3.87e-14
Invar	HFC-125	secondary	6.44e-13	3.26e-14
Ferrite mfg.	HFC-125	secondary	6.29e-13	3.19e-14
Invar	Propionaldehyde	secondary	5.32e-14	2.69e-15
Steel Prod cold-rolled semi-finished	Propionaldehyde	secondary	5 32e-14	2.69e-15
Ferrite mfg.	Propionaldehyde	secondary	5.19e-14	2.63e-15
Invar	Pentachlorobenzene	secondary	2.77e-14	1.40e-15
Steel Prod cold-rolled semi-finished	Tetrachloroethylene	secondary	2.19e-14	1 11e-15
Invar	Tetrachloroethylene	secondary	2.16e-14	1.09e-15
Ferrite mfg	Tetrachloroethylene	secondary	2.11e-14	1.07e-15
Invar	Acrolein	secondary	9 90e-15	5.01e-16
Steel Prod cold-rolled semi-finished	Benzaldehyde	secondary	4 72e-15	2 39e-16
Invar	Benzaldehyde	secondary	4 69e-15	2.39 c 10
Ferrite mfg	Benzaldehyde	secondary	4 57e-15	2.300 10 2.32e-16
Invar	1 1 1-Trichloroethane	secondary	3 25e-15	1.65e-16
Steel Prod cold-rolled semi-finished	1 1 1-Trichloroethane	secondary	3.21e-15	1.63e-16
Ferrite mfg	1 1 1-Trichloroethane	secondary	3.16e-15	1.60e-16
Inver	Perfluoroethane	secondary	3 1/0 15	1.00e-10
Invar	Pentachlorophenol	secondary	1 240 15	6 280 17
Ferrite mfg	1.2 Dichlorotetrafluoroethane	secondary	3 100 16	1.61e 17
Invor	Havashlaraathana	secondary	1 240 16	6 270 18
Steel Prod. cold rolled comi finished	Havashlaroothana	secondary	1.240-10	6 230 18
Farrite mfg	Heyachloroethana	secondary	1.230-10	6 120 19
Formet mfg	Dishlorodifluoromethere	secondary	1.210-10	2 202 19
Ferrite mfg	CEC 13	secondary	4.720-17	2.396-18
Inver	HCEC 22	secondary	5./4e-1/ 9.74a 10	1.090-18
Farrita mfa	HCFC-22	secondary	0.740-19	4.430-20
renne mig.	ПСГС-22	secondary	ð.54e-19	4.32e-20

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Total Materials Processing			2.28e+02	1.16e+01
Manufacturing Life-cycle Stage				
Japanese Electric Grid	Sulfur dioxide	model/secondary	5.69e+01	2.88e+00
US electric grid	Sulfur dioxide	model/secondary	2.63e+01	1.33e+00
LPG Production	Carbon monoxide	secondary	5.34e+00	2.71e-01
LPG Production	Vanadium	secondary	1.05e+00	5.30e-02
LPG Production	Benzene	secondary	8.57e-01	4.34e-02
LPG Production	Methane	secondary	8.39e-01	4.25e-02
LPG Production	Sulfur oxides	secondary	8.01e-01	4.06e-02
Glass/frit	Fluorides (F-)	primary	5.82e-01	2.95e-02
LPG Production	Nitrogen oxides	secondary	5.73e-01	2.90e-02
CRT tube mfg.	Carbon monoxide	primary	2.12e-01	1.07e-02
LPG Production	Arsenic	secondary	2.06e-01	1.05e-02
LPG Production	Formaldehyde	secondary	1.34e-01	6.79e-03
LPG Production	PM	secondary	1.29e-01	6.53e-03
Natural Gas Prod.	Benzene	secondary	7.57e-02	3.83e-03
Natural Gas Prod.	Carbon monoxide	secondary	6.36e-02	3.22e-03
Natural Gas Prod.	Methane	secondary	5.02e-02	2.54e-03
Glass/frit	Nitrogen oxides	primary	4.41e-02	2.24e-03
Japanese Electric Grid	Nitrogen oxides	model/secondary	4.07e-02	2.06e-03
CRT tube mfg.	Phosphorus (yellow or white)	primary	4.00e-02	2.03e-03
LPG Production	Hydrochloric acid	secondary	3.68e-02	1.86e-03
Japanese Electric Grid	Carbon monoxide	model/secondary	3.50e-02	1.77e-03
Fuel Oil #6 Prod.	Carbon monoxide	secondary	2.62e-02	1.33e-03
Japanese Electric Grid	Vanadium	model/secondary	2.27e-02	1.15e-03
US electric grid	Nitrogen oxides	model/secondary	1.88e-02	9.52e-04
LPG Production	Nitrous oxide	secondary	1.62e-02	8.22e-04
US electric grid	Carbon monoxide	model/secondary	1.61e-02	8.17e-04
Fuel Oil #2 Prod.	Carbon monoxide	secondary	1.56e-02	7.89e-04
Fuel Oil #6 Prod.	Vanadium	secondary	1.08e-02	5.45e-04
Japanese Electric Grid	Arsenic	model/secondary	1.04e-02	5.28e-04
US electric grid	Methane	model/secondary	1.03e-02	5.21e-04
CRT tube mfg.	Sulfur oxides	primary	9.96e-03	5.05e-04
LPG Production	Phosphorus (yellow or white)	secondary	9.81e-03	4.97e-04
Natural Gas Prod.	Nitrogen oxides	secondary	9.32e-03	4.72e-04
US electric grid	Arsenic	model/secondary	8.91e-03	4.51e-04
US electric grid	Hydrochloric acid	model/secondary	7.87e-03	3.99e-04
LPG Production	Fluorides (F-)	secondary	7.50e-03	3.80e-04
LPG Production	Selenium	secondary	7.41e-03	3.76e-04
Japanese Electric Grid	Hydrochloric acid	model/secondary	6.28e-03	3.18e-04
Fuel Oil #6 Prod.	Methane	secondary	5.89e-03	2.98e-04
Glass/frit	Fluorides (F-)	primary	5.82e-03	2.95e-04
Fuel Oil #6 Prod.	Sulfur oxides	secondary	5.47e-03	2.77e-04
Fuel Oil #6 Prod.	Benzene	secondary	4.90e-03	2.48e-04
PWB Mfg.	Formaldehyde	model/secondary	4.44e-03	2.25e-04
Fuel Oil #6 Prod.	Nitrogen oxides	secondary	4.09e-03	2.07e-04
LPG Production	Ammonia	secondary	3.74e-03	1.89e-04
CRT tube mfg.	Fluoride	primary	3.45e-03	1.75e-04
Fuel Oil #2 Prod.	Vanadium	secondary	3.45e-03	1.75e-04
LPG Production	Hydrogen sulfide	secondary	3.07e-03	1.55e-04

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
CRT tube mfg.	Dimethyl Formamide	primary	3.03e-03	1.54e-04
Fuel Oil #2 Prod.	Methane	secondary	2.57e-03	1.30e-04
Fuel Oil #2 Prod.	Benzene	secondary	2.55e-03	1.29e-04
Fuel Oil #2 Prod.	Sulfur oxides	secondary	2.44e-03	1.23e-04
CRT tube mfg.	Nitrogen oxides	primary	2.17e-03	1.10e-04
Japanese Electric Grid	PM-10	model/secondary	2.00e-03	1.01e-04
Glass/frit	Carbon monoxide	primary	2.00e-03	1.01e-04
Fuel Oil #2 Prod.	Nitrogen oxides	secondary	1.76e-03	8.90e-05
Fuel Oil #6 Prod.	Arsenic	secondary	1.72e-03	8.74e-05
CRT tube mfg.	Zinc (elemental)	primary	1.65e-03	8.35e-05
LPG Production	Molybdenum	secondary	1.65e-03	8.34e-05
US electric grid	Selenium	model/secondary	1.48e-03	7.50e-05
Fuel Oil #4 Prod.	Carbon monoxide	secondary	1.40e-03	7.11e-05
Fuel Oil #6 Prod.	Formaldehyde	secondary	1.37e-03	6.93e-05
Japanese Electric Grid	Fluorides (F-)	model/secondary	1.31e-03	6.63e-05
Japanese Electric Grid	Selenium	model/secondary	1.27e-03	6.46e-05
LPG Production	Zinc (elemental)	secondary	1.20e-03	6.08e-05
Japanese Electric Grid	Formaldehyde	model/secondary	1.17e-03	5.92e-05
US electric grid	Vanadium	model/secondary	1.11e-03	5.64e-05
LPG Production	Hydrofluoric acid	secondary	1.00e-03	5.08e-05
Fuel Oil #6 Prod.	PM	secondary	9.24e-04	4.68e-05
US electric grid	PM-10	model/secondary	9.22e-04	4.67e-05
LPG Production	Ethane	secondary	7.76e-04	3.93e-05
LPG Production	Phenol	secondary	7.15e-04	3.62e-05
Fuel Oil #2 Prod.	Arsenic	secondary	6.52e-04	3.30e-05
LPG Production	Pentane	secondary	6.51e-04	3.30e-05
CRT tube mfg.	Toluene	primary	6.41e-04	3.25e-05
Japanese Electric Grid	Zinc (elemental)	model/secondary	6.13e-04	3.11e-05
Glass/frit	Nitrogen oxides	primary	5.33e-04	2.70e-05
LPG Production	Hexane	secondary	4.51e-04	2.28e-05
Fuel Oil #2 Prod.	Formaldehyde	secondary	4.41e-04	2.23e-05
Natural Gas Prod.	PM	secondary	4.14e-04	2.10e-05
Fuel Oil #4 Prod.	Vanadium	secondary	4.03e-04	2.04e-05
Fuel Oil #2 Prod.	PM	secondary	3.96e-04	2.01e-05
Japanese Electric Grid	Antimony	model/secondary	3.23e-04	1.64e-05
Natural Gas Prod.	Arsenic	secondary	3.15e-04	1.59e-05
Fuel Oil #6 Prod.	Hydrochloric acid	secondary	2.97e-04	1.50e-05
LPG Production	Nickel	secondary	2.90e-04	1.47e-05
Natural Gas Prod.	Sulfur oxides	secondary	2.76e-04	1.40e-05
Natural Gas Prod.	Ammonia	secondary	2.69e-04	1.36e-05
Fuel Oil #4 Prod.	Methane	secondary	2.61e-04	1.32e-05
Fuel Oil #4 Prod.	Sulfur oxides	secondary	2.45e-04	1.24e-05
Fuel Oil #4 Prod.	Benzene	secondary	2.41e-04	1.22e-05
LPG Production	PM-10	secondary	2.25e-04	1.14e-05
US electric grid	Hydrofluoric acid	model/secondary	2.15e-04	1.09e-05
LPG Production	Antimony	secondary	2.14e-04	1.08e-05
CRT tube mfg.	Nickel	primary	1.89e-04	9.56e-06
Fuel Oil #4 Prod.	Nitrogen oxides	secondary	1.79e-04	9.09e-06
Japanese Electric Grid	Hydrofluoric acid	model/secondary	1.71e-04	8.69e-06
US electric grid	Formaldehyde	model/secondary	1.55e-04	7.84e-06

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Fuel Oil #6 Prod.	Nitrous oxide	secondary	1.45e-04	7.35e-06
LPG Production	Aluminum (+3)	secondary	1.42e-04	7.17e-06
Japanese Electric Grid	Molybdenum	model/secondary	1.31e-04	6.66e-06
Fuel Oil #2 Prod.	Hydrochloric acid	secondary	1.15e-04	5.85e-06
Japanese Electric Grid	Nitrous oxide	model/secondary	1.12e-04	5.66e-06
US electric grid	Benzene	model/secondary	1.11e-04	5.64e-06
Glass/frit	PM	primary	1.09e-04	5.53e-06
CRT tube mfg.	Molybdenum	primary	1.02e-04	5.18e-06
LPG Production	Chromium (VI)	secondary	1.01e-04	5.11e-06
Fuel Oil #6 Prod.	Phosphorus (yellow or white)	secondary	9.77e-05	4.95e-06
Japanese Electric Grid	Benzene	model/secondary	9.10e-05	4.61e-06
Japanese Electric Grid	Methane	model/secondary	8.13e-05	4.12e-06
Fuel Oil #4 Prod.	Arsenic	secondary	7.04e-05	3.57e-06
Natural Gas Prod.	Vanadium	secondary	6.96e-05	3.53e-06
Natural Gas Prod.	Hydrochloric acid	secondary	6.39e-05	3.24e-06
US electric grid	Phosphorus (yellow or white)	model/secondary	6.27e-05	3.18e-06
Fuel Oil #6 Prod.	Selenium	secondary	6.07e-05	3.07e-06
LPG Production	Nitrate	secondary	6.06e-05	3.07e-06
Fuel Oil #6 Prod.	Fluorides (F-)	secondary	5.83e-05	2.96e-06
Glass/frit	Carbon monoxide	primary	5.59e-05	2.83e-06
US electric grid	Nitrous oxide	model/secondary	5.42e-05	2.75e-06
Fuel Oil #2 Prod.	Nitrous oxide	secondary	5.19e-05	2.63e-06
Fuel Oil #4 Prod.	Formaldehyde	secondary	5.14e-05	2.60e-06
Glass/frit	Sulfur oxides	primary	5.08e-05	2.58e-06
Glass/frit	Lead	primary	4.37e-05	2.21e-06
CRT tube mfg	Copper	primary	4 05e-05	2.05e-06
Fuel Oil #4 Prod	PM	secondary	4 05e-05	2.05e-06
Japanese Electric Grid	Nickel	model/secondary	3 51e-05	1 78e-06
LPG Production	Copper	secondary	3.49e-05	1.77e-06
LPG Production	2-Chloroacetophenone	secondary	3 22e-05	1.63e-06
Fuel Oil #2 Prod	Phosphorus (vellow or white)	secondary	3.21e-05	1.63e-06
LPG Production	Dimethylbenzanthracene	secondary	3.19e-05	1.62e-06
LPG Production	Bromomethane	secondary	3.19e-05	1.61e-06
Fuel Oil #6 Prod	Hydrogen sulfide	secondary	3 18e-05	1.61e-06
Glass/frit	Nitrates/nitrites	nrimary	2.94e-05	1.01e 00
Jananese Electric Grid	Barium	model/secondary	2.94e-05	1.49e-06
US electric grid	Zing (alemental)	model/secondary	2.940-05	1.47e.06
Natural Gas Prod	Formaldehyde	secondary	2.890-05	1.47C-00
I BG Production	Nanhthalene	secondary	2.770-05	1.400-00
CPT tube mfs	Naphthalene Vylana (miyad isamara)	secondary	2.586-05	1.316-00
US alastria grid	Antimony	primary madal/accordamy	2.556-05	1.296-00
US electric grid	Anumony	model/secondary	2.376-05	1.200-06
Fuel Oil #2 Prod.	Fluorides (F-)	secondary	2.346-03	1.180-00
Fuel Oli #2 Prod.	Selenium	secondary	2.536-05	1.18e-06
	Manganese	secondary	2.23e-05	1.13e-06
Fuel Oli #6 Prod.	Ammonia	secondary	2.016-05	1.02e-06
	Barlum	secondary	1.86e-05	9.44e-07
LPG Production	Silicon	secondary	1.69e-05	8.5/e-07
	Cyanide (-1)	secondary	1.67e-05	8.4/e-07
Japanese Electric Grid	Naphthalene	model/secondary	1.63e-05	8.28e-07
LPG Production	Barium cmpds	secondary	1.59e-05	8.04e-07
Fuel Oil #6 Prod.	Molybdenum	secondary	1.56e-05	7.89e-07

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Fluorides (F-)	secondary	1.35e-05	6.85e-07
LPG Production	Lead	secondary	1.24e-05	6.26e-07
Fuel Oil #4 Prod.	Hydrochloric acid	secondary	1.23e-05	6.24e-07
Natural Gas Prod.	Selenium	secondary	1.21e-05	6.15e-07
Natural Gas Prod.	Nitrous oxide	secondary	1.15e-05	5.85e-07
Fuel Oil #2 Prod.	Ammonia	secondary	1.10e-05	5.58e-07
Fuel Oil #2 Prod.	Hydrogen sulfide	secondary	1.01e-05	5.12e-07
Natural Gas Prod.	Zinc (elemental)	secondary	8.50e-06	4.31e-07
Fuel Oil #6 Prod.	Hydrofluoric acid	secondary	8.10e-06	4.10e-07
US electric grid	Molybdenum	model/secondary	7.92e-06	4.01e-07
Japanese Electric Grid	Copper	model/secondary	7.14e-06	3.62e-07
Natural Gas Prod.	Ethane	secondary	6.99e-06	3.54e-07
US electric grid	2-Chloroacetophenone	model/secondary	6.88e-06	3.49e-07
US electric grid	Bromomethane	model/secondary	6.81e-06	3.45e-07
Fuel Oil #6 Prod.	Zinc (elemental)	secondary	6.59e-06	3.34e-07
LPG Production	Carbon disulfide	secondary	5.98e-06	3.03e-07
Natural Gas Prod.	Pentane	secondary	5.86e-06	2.97e-07
Fuel Oil #4 Prod.	Nitrous oxide	secondary	5.75e-06	2.91e-07
Japanese Electric Grid	2-Chloroacetophenone	model/secondary	5.50e-06	2.79e-07
Japanese Electric Grid	Bromomethane	model/secondary	5.44e-06	2.76e-07
Fuel Oil #2 Prod.	Molybdenum	secondary	5.33e-06	2.70e-07
LPG Production	Benzyl chloride	secondary	4.68e-06	2.37e-07
Fuel Oil #6 Prod.	Ethane	secondary	4.43e-06	2.25e-07
Natural Gas Prod.	Hexane	secondary	4.06e-06	2.06e-07
LPG Production	Aluminum (elemental)	secondary	3.87e-06	1.96e-07
Fuel Oil #6 Prod.	Pentane	secondary	3.72e-06	1.88e-07
Fuel Oil #4 Prod.	Phosphorus (yellow or white)	secondary	3.70e-06	1.88e-07
LPG Production	Chloroform	secondary	3.64e-06	1.84e-07
CRT tube mfg.	Manganese	primary	3.60e-06	1.83e-07
US electric grid	Cyanide (-1)	model/secondary	3.58e-06	1.81e-07
Fuel Oil #2 Prod.	Zinc (elemental)	secondary	3.54e-06	1.80e-07
Fuel Oil #2 Prod.	Hydrofluoric acid	secondary	3.15e-06	1.60e-07
CRT tube mfg.	Lead	primary	3.01e-06	1.53e-07
Japanese Electric Grid	Cyanide (-1)	model/secondary	2.86e-06	1.45e-07
Fuel Oil #6 Prod.	Nickel	secondary	2.81e-06	1.42e-07
US electric grid	2,3,7,8-TCDD	model/secondary	2.71e-06	1.37e-07
US electric grid	Nickel	model/secondary	2.60e-06	1.32e-07
Fuel Oil #6 Prod.	Hexane	secondary	2.58e-06	1.31e-07
LPG Production	Cobalt	secondary	2.51e-06	1.27e-07
Fuel Oil #4 Prod.	Selenium	secondary	2.50e-06	1.27e-07
Fuel Oil #4 Prod.	Fluorides (F-)	secondary	2.46e-06	1.25e-07
Fuel Oil #6 Prod.	Phenol	secondary	2.42e-06	1.23e-07
Fuel Oil #6 Prod.	PM-10	secondary	2.33e-06	1.18e-07
Natural Gas Prod.	Molybdenum	secondary	2.31e-06	1.17e-07
Fuel Oil #2 Prod.	Ethane	secondary	2.30e-06	1.17e-07
Japanese Electric Grid	2,3,7,8-TCDD	model/secondary	2.22e-06	1.13e-07
US electric grid	Naphthalene	model/secondary	2.16e-06	1.09e-07
Fuel Oil #2 Prod.	Phenol	secondary	2.01e-06	1.02e-07
LPG Production	Acrolein	secondary	1.94e-06	9.83e-08
Fuel Oil #2 Prod.	Pentane	secondary	1.93e-06	9.79e-08

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
US electric grid	Barium	model/secondary	1.86e-06	9.42e-08
Natural Gas Prod.	Phosphorus (yellow or white)	secondary	1.77e-06	8.96e-08
Natural Gas Prod.	Hydrofluoric acid	secondary	1.74e-06	8.83e-08
LPG Production	Methyl chloride	secondary	1.57e-06	7.96e-08
LPG Production	Beryllium	secondary	1.40e-06	7.11e-08
Fuel Oil #2 Prod.	Hexane	secondary	1.34e-06	6.79e-08
US electric grid	Carbon disulfide	model/secondary	1.28e-06	6.48e-08
Fuel Oil #4 Prod.	Hydrogen sulfide	secondary	1.19e-06	6.00e-08
Fuel Oil #6 Prod.	Antimony	secondary	1.18e-06	6.00e-08
Japanese Electric Grid	Cobalt	model/secondary	1.18e-06	5.99e-08
LPG Production	Methyl hydrazine	secondary	1.14e-06	5.76e-08
Fuel Oil #4 Prod.	Ammonia	secondary	1.02e-06	5.18e-08
Japanese Electric Grid	Carbon disulfide	model/secondary	1.02e-06	5.17e-08
US electric grid	Benzyl chloride	model/secondary	1.00e-06	5.08e-08
Fuel Oil #2 Prod.	Nickel	secondary	9.44e-07	4.78e-08
LPG Production	Acetaldehyde	secondary	8.73e-07	4.42e-08
LPG Production	Propionaldehyde	secondary	8.73e-07	4.42e-08
LPG Production	Mercury	secondary	8.69e-07	4.40e-08
LPG Production	Cadmium	secondary	8.10e-07	4.10e-08
Japanese Electric Grid	Benzyl chloride	model/secondary	8.00e-07	4.05e-08
US electric grid	Chloroform	model/secondary	7.79e-07	3.95e-08
Fuel Oil #2 Prod.	PM-10	secondary	7.42e-07	3.76e-08
US electric grid	Manganese	model/secondary	7.28e-07	3.69e-08
LPG Production	Cadmium cmpds	secondary	6.94e-07	3.52e-08
LPG Production	Di(2-ethylhexyl)phthalate	secondary	6.71e-07	3.40e-08
Glass/frit	PM	primary	6.67e-07	3.38e-08
Japanese Electric Grid	Chromium (VI)	model/secondary	6.39e-07	3.24e-08
Fuel Oil #2 Prod.	Antimony	secondary	6.32e-07	3.20e-08
LPG Production	Toluene	secondary	6.23e-07	3.16e-08
Japanese Electric Grid	Chloroform	model/secondary	6.22e-07	3.15e-08
CRT tube mfg.	Cyanide (-1)	primary	6.06e-07	3.07e-08
Fuel Oil #4 Prod.	Molybdenum	secondary	6.03e-07	3.05e-08
Glass/frit	Zinc (elemental)	primary	5.55e-07	2.81e-08
US electric grid	Chromium (VI)	model/secondary	5.48e-07	2.78e-08
Fuel Oil #6 Prod.	Aluminum (+3)	secondary	4.79e-07	2.43e-08
Natural Gas Prod.	Antimony	secondary	4.44e-07	2.25e-08
US electric grid	Acrolein	model/secondary	4.15e-07	2.10e-08
Fuel Oil #6 Prod.	Chromium (VI)	secondary	4.09e-07	2.07e-08
Fuel Oil #2 Prod.	Aluminum (+3)	secondary	3.97e-07	2.01e-08
LPG Production	1,4-Dichlorobenzene	secondary	3.57e-07	1.81e-08
US electric grid	Copper	model/secondary	3.57e-07	1.81e-08
US electric grid	Methyl chloride	model/secondary	3.36e-07	1.70e-08
Fuel Oil #4 Prod.	Hydrofluoric acid	secondary	3.36e-07	1.70e-08
Japanese Electric Grid	Acrolein	model/secondary	3.31e-07	1.68e-08
Fuel Oil #4 Prod.	Zinc (elemental)	secondary	3.31e-07	1.68e-08
LPG Production	Dimethyl sulfate	secondary	3.21e-07	1.63e-08
Fuel Oil #6 Prod.	Copper	secondary	3.17e-07	1.61e-08
US electric grid	Fluoride	model/secondary	3.12e-07	1.58e-08
LPG Production	Isophorone	secondary	3.08e-07	1.56e-08
Natural Gas Prod.	Dimethylbenzanthracene	secondary	2.87e-07	1.46e-08

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Japanese Electric Grid	Methyl chloride	model/secondary	2.69e-07	1.36e-08
Fuel Oil #6 Prod.	2-Chloroacetophenone	secondary	2.60e-07	1.32e-08
Fuel Oil #6 Prod.	Bromomethane	secondary	2.57e-07	1.30e-08
LPG Production	Methyl ethyl ketone	secondary	2.48e-07	1.26e-08
LPG Production	Tetrachloroethylene	secondary	2.45e-07	1.24e-08
US electric grid	Methyl hydrazine	model/secondary	2.43e-07	1.23e-08
Fuel Oil #6 Prod.	Nitrate	secondary	2.43e-07	1.23e-08
Japanese Electric Grid	Toluene	model/secondary	2.39e-07	1.21e-08
Glass/frit	Chromium	primary	2.20e-07	1.12e-08
Fuel Oil #4 Prod.	Ethane	secondary	2.18e-07	1.10e-08
LPG Production	Methyl methacrylate	secondary	2.12e-07	1.08e-08
US electric grid	Lead	model/secondary	2.04e-07	1.03e-08
Glass/frit	Nickel	primary	1.95e-07	9.89e-09
Japanese Electric Grid	Methyl hydrazine	model/secondary	1.94e-07	9.84e-09
US electric grid	Cobalt	model/secondary	1.94e-07	9.83e-09
US electric grid	Acetaldehyde	model/secondary	1.87e-07	9.47e-09
US electric grid	Propionaldehyde	model/secondary	1.87e-07	9.47e-09
Fuel Oil #4 Prod.	Pentane	secondary	1.83e-07	9.26e-09
Fuel Oil #6 Prod.	Dimethylbenzanthracene	secondary	1.82e-07	9.24e-09
Fuel Oil #6 Prod.	Manganese	secondary	1.79e-07	9.07e-09
LPG Production	1,2-Dichloroethane	secondary	1.77e-07	8.96e-09
Fuel Oil #6 Prod.	Silicon	secondary	1.75e-07	8.88e-09
LPG Production	Bromoform	secondary	1.73e-07	8.79e-09
Fuel Oil #2 Prod.	Nitrate	secondary	1.73e-07	8.75e-09
LPG Production	Chromium (III)	secondary	1.72e-07	8.70e-09
LPG Production	Dichloromethane	secondary	1.67e-07	8.48e-09
Fuel Oil #4 Prod.	Phenol	secondary	1.63e-07	8.26e-09
Fuel Oil #2 Prod.	Chromium (VI)	secondary	1.58e-07	8.02e-09
Fuel Oil #6 Prod.	Naphthalene	secondary	1.56e-07	7.91e-09
Japanese Electric Grid	Acetaldehyde	model/secondary	1.49e-07	7.56e-09
Japanese Electric Grid	Propionaldehyde	model/secondary	1.49e-07	7.56e-09
US electric grid	Di(2-ethylhexyl)phthalate	model/secondary	1.44e-07	7.27e-09
Japanese Electric Grid	Cadmium	model/secondary	1.41e-07	7.14e-09
LPG Production	Chlorobenzene	secondary	1.40e-07	7.10e-09
LPG Production	Aromatic hydrocarbons	secondary	1.40e-07	7.09e-09
Fuel Oil #6 Prod.	Cyanide (-1)	secondary	1.35e-07	6.84e-09
LPG Production	Copper (+1 & +2)	secondary	1.31e-07	6.64e-09
Natural Gas Prod.	Naphthalene	secondary	1.30e-07	6.59e-09
Fuel Oil #4 Prod.	Hexane	secondary	1.27e-07	6.42e-09
Natural Gas Prod.	Nickel	secondary	1.23e-07	6.22e-09
US electric grid	Mercury	model/secondary	1.20e-07	6.07e-09
Japanese Electric Grid	Mercury	model/secondary	1.18e-07	5.98e-09
Japanese Electric Grid	Di(2-ethylhexyl)phthalate	model/secondary	1.15e-07	5.81e-09
Fuel Oil #2 Prod.	Copper	secondary	1.12e-07	5.67e-09
LPG Production	2,4-Dinitrotoluene	secondary	1.11e-07	5.65e-09
Fuel Oil #4 Prod.	Nickel	secondary	1.08e-07	5.46e-09
Fuel Oil #2 Prod.	2-Chloroacetophenone	secondary	1.01e-07	5.11e-09
Fuel Oil #6 Prod.	Lead	secondary	9.99e-08	5.06e-09
Fuel Oil #2 Prod.	Bromomethane	secondary	9.99e-08	5.06e-09
Fuel Oil #6 Prod.	Barium	secondary	9.97e-08	5.05e-09

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
US electric grid	Hexane	model/secondary	9.59e-08	4.86e-09
Fuel Oil #2 Prod.	Dimethylbenzanthracene	secondary	9.48e-08	4.80e-09
Natural Gas Prod.	Chromium (VI)	secondary	9.46e-08	4.79e-09
Fuel Oil #4 Prod.	PM-10	secondary	8.70e-08	4.41e-09
Fuel Oil #2 Prod.	Naphthalene	secondary	7.72e-08	3.91e-09
US electric grid	Cadmium	model/secondary	7.69e-08	3.90e-09
Japanese Electric Grid	Hexane	model/secondary	7.66e-08	3.88e-09
Fuel Oil #2 Prod.	Manganese	secondary	7.00e-08	3.54e-09
US electric grid	Dimethyl sulfate	model/secondary	6.87e-08	3.48e-09
LPG Production	o-xylene	secondary	6.86e-08	3.48e-09
US electric grid	Toluene	model/secondary	6.78e-08	3.44e-09
Natural Gas Prod.	Hydrogen sulfide	secondary	6.73e-08	3.41e-09
US electric grid	Isophorone	model/secondary	6.59e-08	3.34e-09
Fuel Oil #4 Prod.	Antimony	secondary	5.92e-08	3.00e-09
Natural Gas Prod.	Barium	secondary	5.90e-08	2.99e-09
Fuel Oil #2 Prod.	Silicon	secondary	5.58e-08	2.83e-09
Natural Gas Prod.	2-Chloroacetophenone	secondary	5.58e-08	2.83e-09
LPG Production	Ethylbenzene	secondary	5.56e-08	2.82e-09
Natural Gas Prod.	Bromomethane	secondary	5.52e-08	2.80e-09
Fuel Oil #2 Prod.	Barium	secondary	5.49e-08	2.78e-09
Japanese Electric Grid	Dimethyl sulfate	model/secondary	5.49e-08	2.78e-09
Fuel Oil #6 Prod.	Barium cmpds	secondary	5.37e-08	2.72e-09
US electric grid	Methyl ethyl ketone	model/secondary	5.32e-08	2.69e-09
Glass/frit	Barium	primary	5.29e-08	2.68e-09
Natural Gas Prod.	Copper	secondary	5.28e-08	2.67e-09
Japanese Electric Grid	Isophorone	model/secondary	5.26e-08	2.66e-09
Fuel Oil #2 Prod.	Cyanide (-1)	secondary	5.25e-08	2.66e-09
US electric grid	Tetrachloroethylene	model/secondary	5.23e-08	2.65e-09
Fuel Oil #6 Prod.	Carbon disulfide	secondary	4.82e-08	2.44e-09
US electric grid	Methyl methacrylate	model/secondary	4.54e-08	2.30e-09
Fuel Oil #2 Prod.	Barium cmpds	secondary	4.45e-08	2.26e-09
Japanese Electric Grid	Methyl ethyl ketone	model/secondary	4.24e-08	2.15e-09
Japanese Electric Grid	Tetrachloroethylene	model/secondary	4.18e-08	2.12e-09
Fuel Oil #6 Prod.	Aluminum (elemental)	secondary	4.01e-08	2.03e-09
Natural Gas Prod.	Manganese	secondary	3.95e-08	2.00e-09
Fuel Oil #2 Prod.	Lead	secondary	3.88e-08	1.97e-09
US electric grid	1.2-Dichloroethane	model/secondary	3.79e-08	1.92e-09
Fuel Oil #6 Prod.	Benzyl chloride	secondary	3.78e-08	1.91e-09
US electric grid	Bromoform	model/secondary	3.71e-08	1.88e-09
Japanese Electric Grid	Methyl methacrylate	model/secondary	3.63e-08	1.84e-09
US electric grid	Dichloromethane	model/secondary	3.58e-08	1.82e-09
Japanese Electric Grid	Bervllium	model/secondary	3.23e-08	1.64e-09
Fuel Oil #4 Prod.	Aluminum (+3)	secondary	3.23e-08	1.64e-09
US electric grid	Bervllium	model/secondary	3.05e-08	1.54e-09
Japanese Electric Grid	1.2-Dichloroethane	model/secondary	3.02e-08	1.53e-09
US electric grid	Chlorobenzene	model/secondary	3.00e-08	1.52e-09
Japanese Electric Grid	Bromoform	model/secondary	2.96e-08	1.50e-09
Fuel Oil #6 Prod.	Chloroform	secondary	2.94e-08	1.49e-09
Natural Gas Prod.	Cvanide (-1)	secondary	2.90e-08	1.47e-09
Japanese Electric Grid	Dichloromethane	model/secondary	2.86e-08	1 45e-09
- The second street and street		model/secondary	2.000 00	1.100 09

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
LPG Production	Methyl tert-butyl ether	secondary	2.79e-08	1.41e-09
Japanese Electric Grid	Chlorobenzene	model/secondary	2.39e-08	1.21e-09
US electric grid	2,4-Dinitrotoluene	model/secondary	2.38e-08	1.21e-09
LPG Production	Phenanthrene	secondary	2.27e-08	1.15e-09
Natural Gas Prod.	Lead	secondary	2.17e-08	1.10e-09
Fuel Oil #6 Prod.	Cobalt	secondary	2.12e-08	1.08e-09
LPG Production	Styrene	secondary	2.03e-08	1.03e-09
LPG Production	Vinyl acetate	secondary	1.98e-08	1.01e-09
Japanese Electric Grid	2,4-Dinitrotoluene	model/secondary	1.90e-08	9.62e-10
Fuel Oil #2 Prod.	Carbon disulfide	secondary	1.87e-08	9.50e-10
LPG Production	3-Methylcholanthrene	secondary	1.87e-08	9.49e-10
Fuel Oil #4 Prod.	Chromium (VI)	secondary	1.69e-08	8.56e-10
Fuel Oil #6 Prod.	Acrolein	secondary	1.57e-08	7.93e-10
Natural Gas Prod.	Nitrate	secondary	1.50e-08	7.62e-10
Fuel Oil #2 Prod.	Benzyl chloride	secondary	1.47e-08	7.44e-10
Fuel Oil #4 Prod.	Nitrate	secondary	1.47e-08	7.43e-10
Natural Gas Prod.	Phenol	secondary	1.46e-08	7.39e-10
Glass/frit	Copper	primary	1.42e-08	7.20e-10
Japanese Electric Grid	Cumene hydroperoxide	model/secondary	1.34e-08	6.80e-10
Fuel Oil #2 Prod.	Aluminum (elemental)	secondary	1.28e-08	6.47e-10
Fuel Oil #6 Prod.	Methyl chloride	secondary	1.27e-08	6.42e-10
Fuel Oil #4 Prod.	Copper	secondary	1.25e-08	6.32e-10
US electric grid	Ethylbenzene	model/secondary	1.18e-08	5.99e-10
LPG Production	Zinc (+2)	secondary	1.17e-08	5.94e-10
Fuel Oil #2 Prod.	Chloroform	secondary	1.14e-08	5.79e-10
Fuel Oil #6 Prod.	Beryllium	secondary	1.12e-08	5.67e-10
Fuel Oil #4 Prod.	2-Chloroacetophenone	secondary	1.08e-08	5.45e-10
Fuel Oil #4 Prod.	Bromomethane	secondary	1.07e-08	5.40e-10
Japanese Electric Grid	Ethylbenzene	model/secondary	1.04e-08	5.26e-10
Natural Gas Prod.	Carbon disulfide	secondary	1.04e-08	5.25e-10
Fuel Oil #6 Prod.	Methyl hydrazine	secondary	9.18e-09	4.65e-10
Fuel Oil #4 Prod.	Dimethylbenzanthracene	secondary	8.97e-09	4.54e-10
Natural Gas Prod.	Toluene	secondary	8.63e-09	4.37e-10
Natural Gas Prod.	Benzyl chloride	secondary	8.12e-09	4.12e-10
LPG Production	Ethylene dibromide	secondary	8.03e-09	4.07e-10
Fuel Oil #2 Prod.	Cobalt	secondary	7.95e-09	4.03e-10
LPG Production	1,1,1-Trichloroethane	secondary	7.57e-09	3.83e-10
Fuel Oil #4 Prod.	Naphthalene	secondary	7.45e-09	3.77e-10
Fuel Oil #4 Prod.	Manganese	secondary	7.44e-09	3.77e-10
Fuel Oil #6 Prod.	Acetaldehyde	secondary	7.05e-09	3.57e-10
Fuel Oil #6 Prod.	Propionaldehyde	secondary	7.05e-09	3.57e-10
Fuel Oil #6 Prod.	Mercury	secondary	6.94e-09	3.52e-10
Fuel Oil #4 Prod.	Silicon	secondary	6.54e-09	3.31e-10
Fuel Oil #6 Prod.	Cadmium	secondary	6.36e-09	3.22e-10
Natural Gas Prod.	Chloroform	secondary	6.32e-09	3.20e-10
Fuel Oil #2 Prod.	Acrolein	secondary	6.09e-09	3.08e-10
LPG Production	2-Methylnaphthalene	secondary	6.01e-09	3.04e-10
US electric grid	Methyl tert-butyl ether	model/secondary	5.96e-09	3.02e-10
Fuel Oil #4 Prod.	Cyanide (-1)	secondary	5.60e-09	2.83e-10
Fuel Oil #6 Prod.	Di(2-ethylhexyl)phthalate	secondary	5.41e-09	2.74e-10

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
LPG Production	Ethyl Chloride	secondary	5.36e-09	2.72e-10
LPG Production	Chlorine	secondary	5.32e-09	2.70e-10
Japanese Electric Grid	Phenanthrene	model/secondary	5.21e-09	2.64e-10
Fuel Oil #4 Prod.	Barium	secondary	5.08e-09	2.58e-10
Natural Gas Prod.	PM-10	secondary	4.94e-09	2.50e-10
Fuel Oil #2 Prod.	Methyl chloride	secondary	4.93e-09	2.50e-10
Japanese Electric Grid	Methyl tert-butyl ether	model/secondary	4.75e-09	2.40e-10
Natural Gas Prod.	o-xylene	secondary	4.60e-09	2.33e-10
US electric grid	Phenol	model/secondary	4.54e-09	2.30e-10
LPG Production	Cumene	secondary	4.54e-09	2.30e-10
Japanese Electric Grid	Chromium (III)	model/secondary	4.48e-09	2.27e-10
Fuel Oil #2 Prod.	Beryllium	secondary	4.40e-09	2.23e-10
US electric grid	Styrene	model/secondary	4.35e-09	2.20e-10
US electric grid	Vinyl acetate	model/secondary	4.25e-09	2.15e-10
Fuel Oil #6 Prod.	Toluene	secondary	4.19e-09	2.12e-10
Fuel Oil #4 Prod.	Lead	secondary	4.14e-09	2.10e-10
US electric grid	Phenanthrene	model/secondary	4.00e-09	2.03e-10
Japanese Electric Grid	1,1,1-Trichloroethane	model/secondary	3.66e-09	1.85e-10
Japanese Electric Grid	Phenol	model/secondary	3.63e-09	1.84e-10
Fuel Oil #4 Prod.	Barium cmpds	secondary	3.62e-09	1.83e-10
Fuel Oil #2 Prod.	Methyl hydrazine	secondary	3.57e-09	1.81e-10
US electric grid	Xylene (mixed isomers)	model/secondary	3.52e-09	1.78e-10
Japanese Electric Grid	Styrene	model/secondary	3.47e-09	1.76e-10
Japanese Electric Grid	Vinyl acetate	model/secondary	3.38e-09	1.71e-10
Natural Gas Prod.	Acrolein	secondary	3.37e-09	1.71e-10
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	3.22e-09	1.63e-10
US electric grid	Chromium (III)	model/secondary	3.15e-09	1.59e-10
Natural Gas Prod.	Aluminum (+3)	secondary	2.86e-09	1.45e-10
LPG Production	Acetophenone	secondary	2.82e-09	1.43e-10
Japanese Electric Grid	Xylene (mixed isomers)	model/secondary	2.81e-09	1.42e-10
Natural Gas Prod.	Beryllium	secondary	2.80e-09	1.42e-10
Fuel Oil #2 Prod.	Acetaldehyde	secondary	2.74e-09	1.39e-10
Fuel Oil #2 Prod.	Propionaldehyde	secondary	2.74e-09	1.39e-10
Natural Gas Prod.	Methyl chloride	secondary	2.73e-09	1.38e-10
Fuel Oil #2 Prod.	Mercury	secondary	2.72e-09	1.38e-10
LPG Production	Biphenyl	secondary	2.71e-09	1.37e-10
Natural Gas Prod.	Cobalt	secondary	2.64e-09	1.34e-10
Fuel Oil #6 Prod.	Dimethyl sulfate	secondary	2.59e-09	1.31e-10
Fuel Oil #2 Prod.	Cadmium	secondary	2.53e-09	1.28e-10
Fuel Oil #6 Prod.	Isophorone	secondary	2.48e-09	1.26e-10
Fuel Oil #6 Prod.	Cadmium cmpds	secondary	2.35e-09	1.19e-10
LPG Production	Acenaphthylene	secondary	2.14e-09	1.08e-10
Fuel Oil #2 Prod.	Di(2-ethylhexyl)phthalate	secondary	2.11e-09	1.07e-10
Fuel Oil #6 Prod.	1,4-Dichlorobenzene	secondary	2.04e-09	1.03e-10
Fuel Oil #6 Prod.	Methyl ethyl ketone	secondary	2.00e-09	1.02e-10
Fuel Oil #4 Prod.	Carbon disulfide	secondary	2.00e-09	1.01e-10
Fuel Oil #6 Prod.	Tetrachloroethylene	secondary	1.97e-09	1.00e-10
Natural Gas Prod.	Methyl hydrazine	secondary	1.97e-09	1.00e-10
Fuel Oil #2 Prod.	Cadmium cmpds	secondary	1.95e-09	9.87e-11
Fuel Oil #2 Prod.	Toluene	secondary	1.90e-09	9.61e-11

Natural Derivationsecondarysecondary1.78-090.910-11US electric gird1.1.1.Frindhoredhmemodel secondary1.73-098.70e-11US electric girdFullylene dibronidemodel secondary1.71e-098.60e-11Del Ol 16 Prod.Methyl meharrylatesecondary1.71e-098.68e-11Darlel Ol 14 Prod.Methyl meharrylatesecondary1.51e-097.67e-11Natural Gas Prod.Aceatledbylesecondary1.51e-097.67e-11Darlel Ol 14 Prod.Aceatledbylesecondary1.51e-097.58e-11Japanses Electric GridAvenaphhenemodel/secondary1.45e-097.47e-01Parlel Ol 14 Prod.Calmiumsecondary1.45e-097.18e-11Parl Ol 16 Prod.1.2.Del/norednanesecondary1.45e-097.18e-11Parl Ol 16 Prod.Calmiumsecondary1.45e-097.18e-11Parl Ol 16 Prod.Diol/toromethanesecondary1.35e-096.85e-11Japanses Electric GridDiylene dibornidemodel/secondary1.35e-096.35e-11Japanses Electric GridDiol/toromethanesecondary1.35e-096.35e-11Parl Ol 14 Prod.Chloroformsecondary1.26e-096.37e-11Parl Ol 14 Prod.Chloroformsecondary1.26e-096.37e-11Parl Ol 14 Prod.Chloroformsecondary1.26e-096.37e-11Parl Ol 14 Prod.Chloroformsecondary1.26e-096.37e-11Parl Ol 14 Prod.Chlor	Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
LPG Production Acemaphthene secondary 1.78e.00 9.78e.10 US electric grid 1.1.1-frichlooothane model/secondary 1.72e.00 8.78e.11 Pad Oli 46 Prod. Benyl incharylar secondary 1.71e.00 8.68e.11 Pad Oli 44 Prod. Benyl ichiorde secondary 1.51e.00 7.67e.11 Natural Gas Prod. Acetaldehyde secondary 1.51e.00 7.67e.11 Japaness Electric Grid Acetaldehyde secondary 1.40e.00 7.89e.11 Japaness Electric Grid Cadantian secondary 1.40e.00 7.89e.11 Pad Oli 46 Prod. Bronoform secondary 1.40e.00 7.04e.11 Pad Oli 46 Prod. Bronoform secondary 1.30e.00 7.89e.11 Pad Oli 46 Prod. Bronoform secondary 1.32e.00 6.85e.11 Pad Oli 46 Prod. Bronoform secondary 1.32e.00 6.85e.11 Pad Oli 44 Prod. Dichloromethane secondary 1.32e.00 6.87e.11 Pad Oli 44 Prod. Chioroforn <t< td=""><td>Natural Gas Prod.</td><td>Mercury</td><td>secondary</td><td>1.88e-09</td><td>9.54e-11</td></t<>	Natural Gas Prod.	Mercury	secondary	1.88e-09	9.54e-11
Us electric grid 1,1,1-Trichloroenhane model/ascondary 1,72-09 8,77-e11 Vas electric grid Edylene dibromide secondary 1,71e-09 8,68e-11 Fuel Oil #6 Prod. Methyl methacrylate secondary 1,51e-09 7,67e-11 Fuel Oil #4 Prod. Acetaldehyde secondary 1,51e-09 7,67e-11 Japanese Electric Grid Acetaldehyde secondary 1,45e-09 7,87e-11 Japanese Electric Grid Acetalphane secondary 1,45e-09 7,87e-11 Japanese Electric Grid Edylene dibromide model/ascondary 1,45e-09 7,89e-11 Japanese Electric Grid Edylene dibromide model/ascondary 1,35e-09 6,58e-11 Japanese Electric Grid o-xylene model/ascondary 1,25e-09 6,58e-11 Japanese Electric Grid o-xylene model/ascondary 1,25e-09 6,58e-11 Japanese Electric Grid o-xylene model/ascondary 1,25e-09 6,58e-11 Japanese Electric Grid Berno/a Intrinee secondary 1,25e-09 6,58e-11	LPG Production	Acenaphthene	secondary	1.78e-09	9.03e-11
US electric grid Ethylene dihromide model'scondary 1.72e.09 8.70e-11 Fael Oli #6 Prod. Methyl mehacry late secondary 1.51e-09 7.84e-11 Fael Oli #6 Prod. Acetaldehyde secondary 1.51e-09 7.87e-11 Natural Gas Prod. Acetaldehyde secondary 1.51e-09 7.87e-11 Fael Oli #6 Prod. Acetaldehyde secondary 1.45e-09 7.87e-11 Fael Oli #6 Prod. Cadmium secondary 1.43e-09 7.82e-11 Fael Oli #6 Prod. Bromoform secondary 1.49e-09 7.09e-11 Fael Oli #6 Prod. Bromoform secondary 1.37e-09 6.98e-11 Fael Oli #6 Prod. Dichloromethane secondary 1.35e-09 6.85e-11 Japanese Electric Grid Dichloromethane secondary 1.28e-09 6.32e-11 Fael Oli #6 Prod. Dichloromethane secondary 1.28e-09 6.32e-11 Fael Oli #4 Prod. Dichoromethane secondary 1.28e-09 6.32e-11 Fael Oli #4 Prod. Chiorometh	US electric grid	1,1,1-Trichloroethane	model/secondary	1.73e-09	8.77e-11
Fuel Oil # Prod.Methyl metharylatesecondary1.71e-09F.88e-11Natural Gas Prod.Acetaldehydesecondary1.51e-097.87e-11Natural Gas Prod.Acetaldehydesecondary1.51e-097.87e-11Fuel Oil # Prod.Aluminum (elemental)secondary1.51e-097.87e-11Japanese Electric GridAcemaphthenemodel/secondary1.47e-097.44e-11Fuel Oil # Prod.1.2-Dichbreethanesecondary1.47e-097.28e-11Patel Oil # Prod.1.2-Dichbreethanesecondary1.49e-097.08e-11Fuel Oil # Prod.Cadmiumsecondary1.39e-097.08e-11Japanese Electric GridDichoromethanesecondary1.37e-096.95e-11Japanese Electric GridDickle angdssecondary1.28e-096.38e-11Japanese Electric GridNickelprimacy1.28e-096.38e-11Japanese Electric GridSecondary1.27e-096.95e-11Lef Or OnductionBencolo Janthracenesecondary1.27e-096.38e-11Lef Or OnductionChioroformsecondary1.27e-096.37e-11Lef Or OnductionChioroformsecondary1.17e-095.91e-11Lef Or OnductionChioroformsecondary1.17e-095.91e-11Lef Or OnductionChioroformsecondary1.18e-095.81e-11Lef Or OnductionDic-2.491heyh phthalatesecondary1.18e-095.81e-11Lef Or OnductionDic-10.491heyh phthalatesecon	US electric grid	Ethylene dibromide	model/secondary	1.72e-09	8.70e-11
Fuel Olif 4 Prod. Benzyl chloride secondary 1.57e-09 7.94e-11 Natural Gas Prod. Propionaldehyde secondary 1.51e-09 7.67e-11 Natural Gas Prod. Aluminum (clemental) secondary 1.51e-09 7.67e-11 Japanese Electric Grid Aluminum (clemental) secondary 1.47e-09 7.44e-11 Paul Oli # 6 Prod. Bronoform secondary 1.48e-09 7.23e-11 Paul Oli # 6 Prod. Bronoform secondary 1.37e-09 6.37e-11 Paul Oli # 6 Prod. Bronoform secondary 1.37e-09 6.43e-11 Japanese Electric Grid Hylene dibromide model/secondary 1.37e-09 6.43e-11 Japanese Electric Grid Benzo [a]antracene secondary 1.25e-09 6.37e-11 Japanese Electric Grid Chloroform secondary 1.25e-09 6.37e-11 Jeal Oli # A Prod. Chloroform secondary 1.12e-09 6.37e-11 JCP Production Lead empds secondary 1.15e-09 5.79e-11 JCP Oroduction	Fuel Oil #6 Prod.	Methyl methacrylate	secondary	1.71e-09	8.68e-11
Natural Gas Prod. Acetaldehyde secondary 1.51e-09 7.67e-11 Natural Gas Prod. Aluminum (elemental) secondary 1.50e-09 7.58e-11 Japanese Electric Grid Acenaphthen model/secondary 1.47e-09 7.45e-11 Ledi Oli #O Prod. 1.27b-ichloroethane secondary 1.46e-09 7.11e-11 Fuel Oli #O Prod. Cadmium secondary 1.46e-09 7.10e-11 Fuel Oli #O Prod. Cadmium secondary 1.36e-09 7.01e-11 Japanese Electric Grid Eitylene dibromide model/secondary 1.37e-09 6.58e-11 Japanese Electric Grid o-xylenc model/secondary 1.28e-09 6.53e-11 Japanese Electric Grid o-xylenc model/secondary 1.28e-09 6.32e-11 JCP Oroduction Bezo Gajahnhacene secondary 1.28e-09 6.32e-11 LPG Production Choroferm secondary 1.17e-09 5.91e-11 LPG Production Lead empts secondary 1.17e-09 5.91e-11 LPG Production <t< td=""><td>Fuel Oil #4 Prod.</td><td>Benzyl chloride</td><td>secondary</td><td>1.57e-09</td><td>7.94e-11</td></t<>	Fuel Oil #4 Prod.	Benzyl chloride	secondary	1.57e-09	7.94e-11
Natural Gas Prod.Propionaldehydesecondary1.51e-097.67e-11Fuel Oli 44 Prod.Atuminum (elemental)secondary1.47e-097.58e-11Fuel Oli 46 Prod.1.2-Dichlorocthanesecondary1.46e-097.11e-11Fuel Oli 46 Prod.Cadmiunsecondary1.40e-097.11e-11LPG ProductionNickel empdssecondary1.40e-097.03e-11LPG ProductionNickel empdssecondary1.37e-096.87e-11Japanese Electric GridErklytene ditromidemodel/secondary1.35e-096.87e-11Japanese Electric Grid-sylenemodel/secondary1.25e-096.32e-11LPG ProductionBenzojaJanthracenesecondary1.25e-096.32e-11LPG ProductionChioroformsecondary1.26e-096.32e-11LPG ProductionChioroformsecondary1.26e-096.77e-11LPG ProductionLead empdssecondary1.15e-095.91e-11LPG ProductionLead empdssecondary1.15e-095.91e-11Natural Gas Prod.Di/2-etylhyserylphthalatesecondary1.16e-095.90e-11US electric gridChiorobenzenesecondary1.16e-095.30e-11US electric gridChiorobenzenesecondary1.06e-095.30e-11US electric gridChiorobenzenesecondary9.66e-104.32e-11LPG ProductionIndered (1.2,2-dipyenesecondary9.66e-104.32e-11LPG ProductionIndered (1.2,2-dipyene <td< td=""><td>Natural Gas Prod.</td><td>Acetaldehyde</td><td>secondary</td><td>1.51e-09</td><td>7.67e-11</td></td<>	Natural Gas Prod.	Acetaldehyde	secondary	1.51e-09	7.67e-11
Fuel Oil # Prod. Aluminum (elemental) secondary 1.50-09 7.58e-11 Japanese Electric Grid Acenaphthene model/secondary 1.47e-09 7.48e-11 Natural Gas Prod. Cadmium secondary 1.40e-09 7.11e-11 LPG Production Nickel empds secondary 1.40e-09 7.00e-11 Japanese Electric Grid Ethylene dibromide model/secondary 1.37e-09 6.57e-11 Japanese Electric Grid Ethylene dibromide model/secondary 1.32e-09 6.53e-11 Japanese Electric Grid e-xylene model/secondary 1.22e-09 6.32e-11 Glass/Erit Nickel primary 1.22e-09 6.32e-11 Glass/Erit Chioroform secondary 1.26e-09 6.32e-11 LPG Production Chioroform secondary 1.17e-09 5.91e-11 Natural Gas Prod. LAberops Electric Grid model/secondary 1.17e-09 5.91e-11 Natural Gas Prod. LAberops Electric Grid Enhylene Glass/Erit 1.13e-09 5.73e-11 Natural Ga	Natural Gas Prod.	Propionaldehyde	secondary	1.51e-09	7.67e-11
Japanese Electric GridAcemphlenemodel/secondary1.47-097.44-e11Fuel Oil #6 Prod.Cadmiumsecondary1.40e-097.23e-11Fuel Oil #6 Prod.Bromoformsecondary1.40e-097.09e-11Japanese Electric GridEthylene dibromidemodel/secondary1.37e-096.55e-11Japanese Electric GridEthylene dibromidemodel/secondary1.37e-096.55e-11Japanese Electric Grid0-sylenemodel/secondary1.28e-096.43e-11Japanese Electric Grid0-sylenemodel/secondary1.28e-096.43e-11LPG ProductionBetro/ajanthracenesecondary1.22e-096.43e-11LPG OrboductionChioroformsecondary1.22e-096.07e-11LPG ProductionChioroformsecondary1.16e-095.91e-11LPG ProductionChiorofenzenesecondary1.16e-095.91e-11LPG ProductionChiorobenzenesecondary1.16e-095.38e-11LPG Off Prod.Chiorobenzenesecondary1.16e-095.38e-11Fuel Oil #2 Prod.1.44Dichiorobenzenesecondary1.06e-095.38e-11Fuel Oil #2 Prod.1.44Dichiorobenzenesecondary9.66e-104.89e-11LPG ProductionIndeno(1,2,3-cd)pyrenesecondary9.66e-104.89e-11LPG ProductionIndeno(1,2,3-cd)pyrenesecondary9.66e-104.89e-11LPG ProductionHordyl ethyl teknesecondary9.66e-104.58e-11LPG Production<	Fuel Oil #4 Prod.	Aluminum (elemental)	secondary	1.50e-09	7.58e-11
Fuel Oil #6 Prod.L2-Dichloroeltanesecondary1.43e-097.23e-11Natural Gas Prod.Cadmiumsecondary1.40e-097.11e-11LPG Prod.Bromoformsecondary1.39e-097.04e-11Japances Electric GridEtylene ditromidemodel'secondary1.37e-096.55e-11Japances Electric Grido-xylenemodel'secondary1.28e-096.55e-11Japances Electric Grido-xylenemodel'secondary1.27e-096.33e-11Japances Electric Grido-xylenesecondary1.27e-096.37e-11LPG ProductionBenzo[ajnthracenesecondary1.20e-096.37e-11LPG ProductionChivoroformsecondary1.20e-096.37e-11Vattori GridLead empdssecondary1.17e-095.90e-11UPG ProductionLead empdssecondary1.17e-095.90e-11US electric gridFibyl Chloridemodel/secondary1.15e-095.38e-11Fuel Oil #2 Prod.L'Alchorobenzenesecondary1.06e-095.38e-11US electric gridDimethyl sulfatesecondary9.06e-104.59e-11US electric gridInden(1.2, 3c-0)pyrenesecondary9.06e-104.59e-11US electric gridCumenemodel/secondary9.16e-104.56e-11Fuel Oil #2 Prod.Lophoronesecondary9.66e-104.56e-11Fuel Oil #2 Prod.Inden(1.2, 3c-0)pyrenesecondary7.76e-103.58e-11Fuel Oil #2 Prod.Edyloramitenesec	Japanese Electric Grid	Acenaphthene	model/secondary	1.47e-09	7.44e-11
Natural Gas Prod. Cadmium secondary 1.40e-09 7.11e-11 Fuel Oil #6 Prod. Bromoform secondary 1.39e-09 7.09e-11 LPG Production Nickel empds secondary 1.37e-09 6.95e-11 Japanese Electric Grid Dichlocromethane secondary 1.28e-09 6.83e-11 Japanese Electric Grid o-xylene model'secondary 1.28e-09 6.33e-11 Japanese Electric Grid o-xylene secondary 1.25e-09 6.33e-11 Japanese Electric Grid Chloroform secondary 1.25e-09 6.33e-11 LPG Production Leanzolglanthraene secondary 1.25e-09 6.37e-11 LPG Production Lead cmpds secondary 1.16e-09 5.99e-11 Natural Gas Prod. Di(2-ethylhexyl)phthalate secondary 1.16e-09 5.99e-11 US electric grid Cumene secondary 1.13e-09 5.88e-11 Fuel Oil # Drod. Lohorbon-zne secondary 9.16e-10 4.92e-11 US electric grid Cumene <	Fuel Oil #6 Prod.	1,2-Dichloroethane	secondary	1.43e-09	7.23e-11
Fuel Oil #6 Prod. Bromoform secondary 1.490-P0 7.09e-11 LPG Production Nickel empds secondary 1.37e-00 7.04e-11 Japanese Electric Grid Ethylene dibromide model/secondary 1.37e-00 6.58e-11 Japanese Electric Grid o-xylene model/secondary 1.28e-09 6.35e-11 Japanese Electric Grid o-xylene secondary 1.27e-09 6.33e-11 LPG Production Benzo[ajanthracene secondary 1.26e-09 6.32e-11 LPG Production Chrysene secondary 1.20e-09 6.17e-11 LPG Production Lead empds secondary 1.16e-09 5.91e-11 Natural Gas Prod. Di(2-ethylhexyl)phthalate secondary 1.16e-09 5.81e-11 Sel Oli #6 Prod. Diforobenzene secondary 1.16e-09 5.81e-11 Sel Oli #2 Prod. Dimethyl sulfate secondary 1.01e-09 5.81e-11 Sel Oli #2 Prod. Dimethyl sulfate secondary 9.06e-10 4.70e-11 Sel Oli #2 Prod. Soph	Natural Gas Prod.	Cadmium	secondary	1.40e-09	7.11e-11
I.PG ProductionNickel empdssecondary1.39e-097.04e-11Japanese Electric GridEthylene dibromidemodel/secondary1.35e-096.59e-11Japanese Electric Grido-xylenemodel/secondary1.28e-096.50e-11Glass/fritNickelprimary1.27e-006.43e-11LPG ProductionBenzo[Janthracenesecondary1.22e-096.17e-11LPG ProductionChloroformsecondary1.22e-096.17e-11LPG ProductionLead empdssecondary1.17e-095.91e-11DFG ProductionLead empdssecondary1.16e-095.90e-11Natural Gas Prod.Di(2-ethylhexyl)phthalatesecondary1.15e-095.81e-11Puel Oil # Prod.Chlorobenzenesecondary1.13e-095.73e-11Fuel Oil # 2 Prod.1,4-Dichlorobenzenesecondary1.01e-095.73e-11Fuel Oil # 2 Prod.1,4-Dichlorobenzenesecondary9.70e-104.98e-11Fuel Oil # 2 Prod.1.01e-01sophoronesecondary9.70e-104.98e-11Fuel Oil # 2 Prod.Indemo(1,2,3-cd)pyrenesecondary9.70e-104.98e-11Japanese Electric GridEthyl Chloridemodel/secondary9.76e-104.98e-11Japanese Electric GridEthyl Chloridemodel/secondary9.76e-104.98e-11Japanese Electric GridBenzo[Jajhthracenesecondary8.99e-104.75e-11Japanese Electric GridBenzo[Jajhthracenesecondary8.99e-104.75e-11 </td <td>Fuel Oil #6 Prod.</td> <td>Bromoform</td> <td>secondary</td> <td>1.40e-09</td> <td>7.09e-11</td>	Fuel Oil #6 Prod.	Bromoform	secondary	1.40e-09	7.09e-11
Japanese Electric GridEthylene dibromidemodel/secondary1.37e-096.95e-11Fuel Oli #6 Prod.Dichloromethanesecondary1.32e-096.50e-11Glass/fritNickelprimary1.27e-096.43e-11LPG ProductionBenzo[a]uthracenesecondary1.22e-096.17e-11LPG ProductionChloroformsecondary1.22e-096.07e-11LPG ProductionChloroformsecondary1.26e-096.07e-11LPG ProductionLead empdssecondary1.17e-095.91e-11LPG ProductionLead empdssecondary1.17e-095.91e-11US electric gridEthyl Chloridemodel/secondary1.15e-095.81e-11Fuel Oli #2 Prod.1.4-Dichlorobenzenesecondary1.06e-095.38e-11Fuel Oli #2 Prod.1.4-Dichlorobenzenesecondary1.01e-095.10e-11US electric gridCumenemodel/secondary9.06e-104.89e-11LPG ProductionInden(1,2,3-cd)pyrenesecondary9.29e-104.76e-11Japanese Electric GridEnhyl Chloridemodel/secondary9.29e-104.76e-11Japanese Electric GridBenzo[a]uthracenemodel/secondary9.29e-104.56e-11Fuel Oli #2 Prod.Cobaltsecondary7.76e-103.89e-11Japanese Electric GridBenzo[a]uthracenemodel/secondary7.76e-103.89e-11Fuel Oli #2 Prod.Cobaltsecondary7.76e-103.89e-11Fuel Oli #2 Prod.Benzo[a]uthrac	LPG Production	Nickel cmpds	secondary	1.39e-09	7.04e-11
Fuel Oil #6 Prod. Dichloromethane secondary 1.35e-09 6.85e-11 Japanese Electric Grid o-xylene model/secondary 1.27e-09 6.43e-11 LPG Production Benzo[a]anthracene secondary 1.25e-09 6.32e-11 LPG Production Chioroform secondary 1.26e-09 6.07e-11 LPG Production Chrysene secondary 1.17e-09 5.91e-11 Natural Gas Prod. Di(2-etty)hexyl)phthalate secondary 1.15e-09 5.81e-11 US electric grid Ethyl Chloride model/secondary 1.15e-09 5.81e-11 US electric grid Chiorobenzene secondary 1.15e-09 5.81e-11 US electric grid Cumene model/secondary 9.71e-10 4.92e-11 Vel Oil #2 Prod. Dimethyl sulfate secondary 9.66e-10 4.8e-11 US electric grid Cumene model/secondary 9.16e-10 4.64e-11 Japanese Electric Grid Ethyl Chloride model/secondary 9.06e-10 4.58e-11 Puel Oil #2 Prod. <t< td=""><td>Japanese Electric Grid</td><td>Ethylene dibromide</td><td>model/secondary</td><td>1.37e-09</td><td>6.95e-11</td></t<>	Japanese Electric Grid	Ethylene dibromide	model/secondary	1.37e-09	6.95e-11
Japanese Electric Grid o-xylene model/secondary 1.28e-09 6.50e-11 Glass/frit Nickel primary 1.27e-09 6.43e-11 LPG Production Benzo[a]anthracene secondary 1.22e-09 6.17e-11 LPG Production Choroform secondary 1.26e-09 6.07e-11 LPG Production Choroform secondary 1.16e-09 5.90e-11 LPG Production Led empds secondary 1.16e-09 5.90e-11 US electric grid Ethyl Chloride model/secondary 1.13e-09 5.38e-11 Fuel Oil #2 Prod. 1.4De/Dihlorobenzene secondary 1.01e-09 5.38e-11 Fuel Oil #2 Prod. Dimethyl sulfate secondary 9.01e-09 5.38e-11 Fuel Oil #2 Prod. Dimethyl sulfate secondary 9.01e-10 4.92e-11 LPG Production Indeno(1.2.3-cd)pyrene secondary 9.06e-10 4.89e-11 LPG Production Indeno(1.2.3-cd)pyrene secondary 9.06e-10 4.56e-11 LPG Oil #2 Prod. Cobalt secondar	Fuel Oil #6 Prod.	Dichloromethane	secondary	1.35e-09	6.85e-11
Glass/fritNickelprimary1.27c-096.43c-11LPG ProductionBerzo[a]anthracenesecondary1.22c-096.37c-11LPG ProductionChloroformsecondary1.20c-096.07c-11LPG ProductionLead empdssecondary1.17c-095.91c-11LPG ProductionLead empdssecondary1.16c-095.91c-11US electric gridEthyl Chloridemodel/secondary1.15c-095.73c-11Fuel Oil #2 Prod.Dinethyl sulfatesecondary1.06c-095.38c-11Fuel Oil #2 Prod.Dinethyl sulfatesecondary9.71c-104.92c-11US electric gridCumenemodel/secondary9.71c-104.92c-11Fuel Oil #2 Prod.Lophoronesecondary9.71c-104.92c-11US electric gridCumenemodel/secondary9.71c-104.92c-11Fuel Oil #2 Prod.Lophoronesecondary9.66c-104.89c-11Japanese Electric GridEthyl Chloridemodel/secondary9.16c-104.64c-11Fuel Oil #0 Prod.Cobaltsecondary8.90c-104.56c-11Japanese Electric GridBenzo[a]anthracenemodel/secondary8.90c-104.56c-11Japanese Electric GridBenzo[a]anthracenesecondary7.70c-103.89c-11Japanese Electric GridBenzo[a]anthracenesecondary7.70c-103.89c-11LPG ProductionBenzo[b]k[Horanthenesecondary6.66c-103.37c-11Japanese Electric GridBenzo[b]a]anthracene <td>Japanese Electric Grid</td> <td>o-xylene</td> <td>model/secondary</td> <td>1.28e-09</td> <td>6.50e-11</td>	Japanese Electric Grid	o-xylene	model/secondary	1.28e-09	6.50e-11
LPG ProductionBenzo[a]anthracenesecondary1.25e-096.32e-11Fuel Oil #4 Prod.Chloroformsecondary1.20e-096.07e-11LPG ProductionChrysenesecondary1.17e-095.91e-11Natural Gas Prod.Di(2-etty)hrey)]phthalatesecondary1.16e-095.90e-11US electric gridEthyl Chloridemodel/secondary1.15e-095.81e-11Fuel Oil #2 Prod.Chlorobenzenesecondary1.06e-095.38e-11Fuel Oil #2 Prod.I.4-Dichlorobenzenesecondary1.06e-095.38e-11Fuel Oil #2 Prod.Dimethyl sulfatesecondary9.71e-104.92e-11US electric gridCurmenemodel/secondary9.71e-104.92e-11LPG ProductionIndeno(1,2,3-ed)pyrenesecondary9.20e-104.70e-11LPG ProductionIndeno(1,2,3-ed)pyrenesecondary9.20e-104.70e-11LPG ProductionIndeno(1,2,3-ed)pyrenesecondary8.62e-104.37e-11Japanese Electric GridEthyl Chloridesecondary8.62e-104.37e-11Japanese Electric GridBenzo[a]anthracenemodel/secondary8.02e-104.58e-11Fuel Oil #2 Prod.Cobaltsecondary7.79e-103.95e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.36e-103.37e-11Japanese Electric GridBenzo[a]anthracenesecondary7.36e-103.37e-11LPG ProductionMethyl ethyl ketonesecondary7.36e-103.37e-11 <td< td=""><td>Glass/frit</td><td>Nickel</td><td>primary</td><td>1.27e-09</td><td>6.43e-11</td></td<>	Glass/frit	Nickel	primary	1.27e-09	6.43e-11
Fuel Oil #4 Prod.Chloroformsecondary1.22e-096.17e-11LPG ProductionChrysnesecondary1.17e-095.91e-11LPG ProductionLead empdssecondary1.16e-095.91e-11Natural Gas Prod.Di(2-eth)lhexyl)phthalatesecondary1.15e-095.91e-11US electric gridEthyl Chloridemodel/secondary1.15e-095.81e-11Fuel Oil #0 Prod.Chlorobenzenesecondary1.13e-095.38e-11Fuel Oil #2 Prod.1.4-Dichlorobenzenesecondary1.01e-095.10e-11US electric gridCumenemodel/secondary9.71e-104.92e-11Fuel Oil #2 Prod.Isophoronesecondary9.66e-104.89e-11LPG ProductionIndeno(1,2,3-cd)pyrenesecondary9.66e-104.56e-11Fuel Oil #0 Frod.2,4-Dinitrotoluenesecondary8.02e-104.37e-11Japanese Electric GridEthyl Chloridemodel/secondary8.02e-104.37e-11Fuel Oil #2 Prod.Methyl ethonesecondary8.02e-104.37e-11Japanese Electric GridBenzo[a]anthracenemodel/secondary8.02e-104.37e-11Fuel Oil #2 Prod.Methyl Hyl ketonesecondary7.36e-103.37e-11Fuel Oil #2 Prod.Methyl ethyl ketonesecondary7.36e-103.37e-11Fuel Oil #2 Prod.Methyl methacrylatesecondary6.36e-103.37e-11Fuel Oil #2 Prod.Chornium (III)secondary6.66e-103.37e-11Fuel Oil #2	LPG Production	Benzo[a]anthracene	secondary	1.25e-09	6.32e-11
LPG ProductionChrysenesecondary1.20e-096.07e-11LPG ProductionLad cmpdssecondary1.17e-095.91e-11Natural Gas Prod.Di(2-ethylhexyl)phthalatesecondary1.16e-095.90e-11US electric gridEthyl Chloridemodel/secondary1.15e-095.81e-11Fuel Oil #0 Prod.Chlorobenzenesecondary1.06e-095.38e-11Fuel Oil #2 Prod.Dimethyl sulfatesecondary1.01e-095.10e-11US electric gridCumenemodel/secondary9.71e-104.92e-11Fuel Oil #2 Prod.Isophoronesecondary9.66e-104.89e-11LPG ProductionIndeno(1,2,3-ed)pyrenesecondary9.20e-104.70e-11Japanese Electric GridEthyl Chloridemodel/secondary9.16e-104.56e-11Fuel Oil #4 Prod.Cobaltsecondary8.00e-104.56e-11Fuel Oil #4 Prod.Cobaltsecondary8.00e-104.56e-11Fuel Oil #2 Prod.Methyl ethyl ketonesecondary7.70e-103.58e-11Fuel Oil #2 Prod.Methyl ethyl ketonesecondary7.70e-103.58e-11Fuel Oil #2 Prod.Methyl ethyl ketonesecondary7.67e-103.58e-11Fuel Oil #2 Prod.Chronium (III)secondary6.66e-103.57e-11Fuel Oil #2 Prod.Chronium (III)secondary6.66e-103.57e-11Fuel Oil #2 Prod.Acroleinsecondary6.66e-103.57e-11Fuel Oil #2 Prod.Acroleinsecon	Fuel Oil #4 Prod.	Chloroform	secondary	1.22e-09	6.17e-11
LPG ProductionLead empdssecondary1.17e-095.91e-11Natural Gas Prod.Di(2-ethylhexyl)phthalatesecondary1.16e-095.90e-11US electric gridEthyl Chloridemodel/secondary1.15e-095.37e-11Fuel Oil #O Prod.Chlorobenzenesecondary1.01e-095.38e-11Fuel Oil #2 Prod.1.4-Dichlorobenzenesecondary1.01e-095.10e-11US electric gridCumenemodel/secondary9.71e-104.92e-11LPel Oil #2 Prod.Isophoronesecondary9.66e-104.89e-11LPG ProductionIndeno(1,2,3-ed)pyrenesecondary9.29e-104.70e-11Japanese Electric GridEthyl Chloridemodel/secondary9.16e-104.64e-11Fuel Oil #2 Prod.2.4-Dinitrotluenesecondary8.99e-104.56e-11Fuel Oil #4 Prod.Cobaltsecondary8.00e-104.05e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.79e-103.58e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.67e-103.58e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.67e-103.58e-11Fuel Oil #2 Prod.Chromium (III)secondary6.66e-103.37e-11LPG ProductionBenzo[ajprenesecondary6.66e-103.37e-11Fuel Oil #2 Prod.Chromium (III)secondary6.06e-103.37e-11Fuel Oil #2 Prod.Actoleinsecondary6.06e-103.37e-11Fuel Oil #2 Prod.Ac	LPG Production	Chrysene	secondary	1.20e-09	6.07e-11
Natural Gas Prod.Di(2-ethylhexyl)phthalatesecondary1.16e-095.90e-11US electric gridEthyl Chloridemodel/secondary1.15e-095.81e-11Fuel Oil #0 Prod.Chlorobenzenesecondary1.06e-095.38e-11Fuel Oil #2 Prod.1.4-Dichlorobenzenesecondary1.01e-095.10e-11US electric gridCumenemodel/secondary9.71e-104.92e-11Fuel Oil #2 Prod.Isophoronesecondary9.29e-104.70e-11Japanese Electric GridIndeno(1,2,3-cd)pyrenesecondary9.29e-104.70e-11Japanese Electric GridEthyl Chloridemodel/secondary9.16e-104.64e-11Fuel Oil #0 Prod.2.4-Dinitrotoluenesecondary8.09e-104.65e-11Fuel Oil #4 Prod.Cobaltsecondary8.09e-104.56e-11Fuel Oil #2 Prod.Benzo[a]anthracenemodel/secondary8.09e-104.56e-11Fuel Oil #2 Prod.Methyl ethyl ktonesecondary7.07e-103.58e-11Fuel Oil #2 Prod.Methyl ethyl ktonesecondary7.07e-103.58e-11Fuel Oil #2 Prod.Methyl ethyl Romanthenesecondary7.07e-103.58e-11Fuel Oil #2 Prod.Methyl ethyl Retoresecondary6.06e-103.37e-11Fuel Oil #2 Prod.Methyl methacrylatesecondary6.06e-103.37e-11Fuel Oil #2 Prod.Methyl methacrylatesecondary6.06e-103.37e-11Fuel Oil #2 Prod.Acroleinsocolary6.06e-103.37e-11	LPG Production	Lead cmpds	secondary	1.17e-09	5.91e-11
US electric gridEthyl Chloridemodel/secondary1.15e-095.81e-11Fuel Oil #O Prod.1.4-Dichlorobenzenesecondary1.13e-095.73e-11Fuel Oil #2 Prod.1.4-Dichlorobenzenesecondary1.01e-095.10e-11US electric gridCumenemodel/secondary9.71e-104.92e-11Fuel Oil #2 Prod.Isophoronesecondary9.66e-104.89e-11LPG ProductionIndeno(1,2,3-cd)pyrenesecondary9.29e-104.70e-11Japanese Electric GridEthyl Chloridemodel/secondary9.9e-104.64e-11Fuel Oil #4 Prod.2.4-Dinitrotoluenesecondary8.99e-104.64e-11Fuel Oil #4 Prod.Cobaltsecondary8.00e-104.37e-11Japanese Electric GridBenzo[a]anthracenemodel/secondary8.00e-104.37e-11Japanese Electric GridBenzo[a]anthracenemodel/secondary7.67e-103.95e-11Fuel Oil #2 Prod.Methyl ethyl ketonesecondary7.67e-103.95e-11Fuel Oil #2 Prod.Methyl ethyl ketonesecondary7.36e-103.73e-11LPG ProductionBenzo[a]yprenesecondary7.36e-103.52e-11LPG ProductionBenzo[a]yprenesecondary6.66e-103.35e-11LPG ProductionBenzo[a]yprenesecondary6.66e-103.35e-11LPG ProductionBenzo[a]yprenesecondary6.66e-103.35e-11LPG ProductionFluorenemodel/secondary6.66e-103.35e-11LPG Pr	Natural Gas Prod.	Di(2-ethylhexyl)phthalate	secondary	1.16e-09	5.90e-11
Fuel Oil #6 Prod.Chlorobenzenesecondary1.13e-095.73e-11Fuel Oil #2 Prod.1,4-Dichlorobenzenesecondary1.06e-095.38e-11Fuel Oil #2 Prod.Dimethyl sulfatesecondary9.71e-104.92e-11US electric gridCumenemodel/secondary9.76e-104.89e-11LPG ProductionIndeno(1,2,3-cd)pyrenesecondary9.29e-104.70e-11Japanese Electric GridEthyl Chloridemodel/secondary9.16e-104.64e-11Fuel Oil #4 Prod.2,4-Dinitrotoluenesecondary8.09e-104.56e-11Fuel Oil #4 Prod.Cobaltsecondary8.00e-104.05e-11Japanese Electric GridBenzo[a]anthracenemodel/secondary8.00e-104.05e-11Fuel Oil #2 Prod.Cobaltsecondary7.79e-103.95e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.79e-103.95e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.76e-103.89e-11LPG ProductionBenzo[b]k,k[fluoranthenesecondary7.07e-103.58e-11LPG ProductionBenzo[a]pyrenesecondary6.66e-103.37e-11LPG ProductionBenzo[a]pyrenesecondary6.66e-103.37e-11LPG ProductionBenzo[a]pyrenesecondary6.66e-103.35e-11LPG ProductionFluorenesecondary6.66e-103.35e-11LPG ProductionFluorenesecondary6.66e-103.35e-11LPG ProductionFluorene <t< td=""><td>US electric grid</td><td>Ethyl Chloride</td><td>model/secondary</td><td>1.15e-09</td><td>5.81e-11</td></t<>	US electric grid	Ethyl Chloride	model/secondary	1.15e-09	5.81e-11
Fuel Oil #2 Prod.1,4-Dichlorobenzenesecondary1.06e-095.38e-11Fuel Oil #2 Prod.Dimethyl sulfatesecondary1.01e-095.10e-11US electric gridCumenemodel/secondary9.71e-104.92e-11Fuel Oil #2 Prod.Isophoronesecondary9.66e-104.89e-11JPG ProductionIndeno(1,2,3-cd)pyrenesecondary9.29e-104.70e-11Japanese Electric GridEthyl Chloridemodel/secondary9.16e-104.64e-11Fuel Oil #6 Prod.2,4-Dinitrotoluenesecondary8.62e-104.37e-11Japanese Electric GridBenzo[a]anthracenemodel/secondary8.62e-104.37e-11Japanese Electric GridBenzo[a]anthracenemodel/secondary8.62e-104.37e-11Fuel Oil #2 Prod.Cobaltsecondary7.7e-103.95e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.7e-103.58e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.07e-103.58e-11LPG ProductionBenzo[a]pyrenesecondary7.07e-103.58e-11LPG ProductionBenzo[a]pyrenesecondary6.66e-103.37e-11LPG ProductionBenzo[a]pyrenesecondary6.66e-103.37e-11LPG ProductionBenzo[a]pyrenesecondary6.66e-103.37e-11LPG ProductionBenzo[a]pyrenesecondary6.66e-103.37e-11LPG ProductionFluorenesecondary6.66e-103.37e-11LPG ProductionF	Fuel Oil #6 Prod.	Chlorobenzene	secondary	1.13e-09	5.73e-11
Fuel Oil #2 Prod.Dimethyl sulfatesecondary1.01e-095.10e-11US electric gridCumenemodel/secondary9.71e-104.92e-11Fuel Oil #2 Prod.Isophoronesecondary9.66e-104.89e-11LPG ProductionIndeno(1,2,3-ed)pyrenesecondary9.29e-104.70e-11Japanese Electric GridEthyl Chloridemodel/secondary9.16e-104.64e-11Fuel Oil #4 Prod.Cobaltsecondary8.99e-104.56e-11Fuel Oil #4 Prod.Cobaltsecondary8.00e-104.05e-11Fuel Oil #2 Prod.Methyl ethyl ketonesecondary7.79e-103.95e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.67e-103.89e-11LPG ProductionBenzo[ajanthracenesecondary7.07e-103.58e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.07e-103.58e-11Fuel Oil #2 Prod.Chromium (III)secondary7.07e-103.58e-11Fuel Oil #6 Prod.Chromium (III)secondary6.66e-103.37e-11LPG ProductionBenzo[ajpyrenesecondary6.66e-103.37e-11Fuel Oil #4 Prod.Acetophenonemodel/secondary6.06e-103.32e-11LPG ProductionFluorenesecondary6.66e-103.37e-11LPG ProductionFluorenesecondary6.06e-103.32e-11LVG Di #2 Prod.Acetophenonemodel/secondary6.06e-103.32e-11US electric gridAcetophenonemodel/se	Fuel Oil #2 Prod.	1,4-Dichlorobenzene	secondary	1.06e-09	5.38e-11
US electric gridCumenemodel/secondary9.71e-104.92e-11Fuel Oil #2 Prod.Isophoronesecondary9.66e-104.89e-11LPG ProductionInden(1,2,3-cd)pyrenesecondary9.29e-104.70e-11Japanese Electric GridEthyl Chloridemodel/secondary9.16e-104.64e-11Fuel Oil #6 Prod.2,4-Dinitrotoluenesecondary8.02e-104.56e-11Fuel Oil #4 Prod.Cobaltsecondary8.02e-104.37e-11Japanese Electric GridBenzo[a]anthracenemodel/secondary8.00e-104.05e-11Fuel Oil #2 Prod.Methyl ethyl ketonesecondary7.79e-103.95e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.67e-103.89e-11LPG ProductionMercury compoundssecondary7.36e-103.73e-11LPG ProductionMercury compoundssecondary6.96e-103.52e-11Fuel Oil #2 Prod.Chromium (III)secondary6.66e-103.37e-11LPG ProductionBenzo[a]pyrenesecondary6.66e-103.37e-11LPG ProductionFluorenesecondary6.60e-103.37e-11LPG ProductionFluorenesecondary6.04e-103.06e-11Stel Oil #4 Prod.Acctophenonemodel/secondary6.04e-103.29e-11US electric gridAcetophenonemodel/secondary5.79e-102.93e-11US electric gridBiphenylmodel/secondary5.71e-102.89e-11Natural Gas Prod.Pyrene <td>Fuel Oil #2 Prod.</td> <td>Dimethyl sulfate</td> <td>secondary</td> <td>1.01e-09</td> <td>5.10e-11</td>	Fuel Oil #2 Prod.	Dimethyl sulfate	secondary	1.01e-09	5.10e-11
Fuel Oil #2 Prod.Isophoronesecondary9.66e-104.89e-11LPG ProductionIndeno(1,2,3-cd)pyrenesecondary9.29e-104.70e-11Japanese Electric GridEthyl Chloridemodel/secondary9.16e-104.64e-11Fuel Oil #6 Prod.2,4-Dinitrotoluenesecondary8.99e-104.56e-11Fuel Oil #4 Prod.Cobaltsecondary8.62e-104.37e-11Japanese Electric GridBenzo[a]anthracenemodel/secondary8.00e-104.05e-11Fuel Oil #2 Prod.Methyl ethyl ketonesecondary7.79e-103.58e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.67e-103.89e-11LPG ProductionBenzo[b,j,k]fluoranthenesecondary7.07e-103.58e-11Fuel Oil #6 Prod.Chromium (III)secondary6.96e-103.52e-11LPG ProductionBenzo[a]prenesecondary6.60e-103.37e-11LPG ProductionBenzo[a]prenesecondary6.60e-103.37e-11LPG ProductionFluorenesecondary6.60e-103.37e-11LPG ProductionFluorenesecondary6.60e-103.37e-11LPG ProductionFluorenesecondary6.60e-103.37e-11LPG ProductionFluorenesecondary6.49e-103.29e-11US electric gridActophenonemodel/secondary6.49e-103.29e-11US electric gridBiphenylmodel/secondary5.79e-102.93e-11LPG ProductionPyrenesecondary<	US electric grid	Cumene	model/secondary	9.71e-10	4.92e-11
LPG ProductionIndeno(1,2,3-cd)pyrenesecondary9.29e-104.70e-11Japanese Electric GridEthyl Chloridemodel/secondary9.16e-104.64e-11Fuel Oil #6 Prod.2,4-Dinitrotoluenesecondary8.99e-104.56e-11Fuel Oil #4 Prod.Cobaltsecondary8.62e-104.37e-11Japanese Electric GridBenzo[a]anthracenemodel/secondary8.00e-104.05e-11Fuel Oil #2 Prod.Methyl ethyl ketonesecondary7.79e-103.95e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.67e-103.89e-11LPG ProductionBenzo[b_j,k]fluoranthenesecondary7.36e-103.73e-11LPG ProductionMercury compoundssecondary7.07e-103.52e-11Fuel Oil #6 Prod.Chromium (III)secondary6.96e-103.52e-11LPG ProductionBenzo[a]pyrenesecondary6.66e-103.37e-11LPG ProductionFluorenesecondary6.60e-103.37e-11LPG ProductionFluorenesecondary6.60e-103.37e-11LPG ProductionFluorenesecondary6.49e-103.29e-11US electric gridAcetophenonemodel/secondary6.49e-103.29e-11US electric gridBiphenylmodel/secondary5.77e-102.89e-11Natural Gas Prod.Dimethyl sulfatesecondary5.77e-102.89e-11Natural Gas Prod.J.2-Dichloroethanesecondary5.57e-102.82e-11Fuel Oil #2 Prod.Be	Fuel Oil #2 Prod.	Isophorone	secondary	9.66e-10	4.89e-11
Japanese Electric GridEthyl Chloridemodel/secondary9.16e-104.64e-11Fuel Oil #6 Prod.2,4-Dinitrotoluenesecondary8.99e-104.56e-11Fuel Oil #4 Prod.Cobaltsecondary8.62e-104.37e-11Japanese Electric GridBenzo[a]anthracenemodel/secondary8.00e-104.05e-11Fuel Oil #2 Prod.Methyl ethyl ketonesecondary7.79e-103.95e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.67e-103.89e-11LPG ProductionBenzo[b,jk]fluoranthenesecondary7.36e-103.73e-11LPG ProductionMercury compoundssecondary6.96e-103.52e-11LPG ProductionBenzo[a]prenesecondary6.83e-103.46e-11Fuel Oil #2 Prod.Chromium (III)secondary6.66e-103.37e-11LPG ProductionBenzo[a]prenesecondary6.66e-103.37e-11LPG ProductionFluorenesecondary6.60e-103.35e-11Fuel Oil #2 Prod.Methyl methacrylatesecondary6.60e-103.37e-11LPG ProductionFluorenesecondary6.49e-103.29e-11US electric gridAcetophenonemodel/secondary6.49e-103.06e-11US electric gridBiphenylmodel/secondary5.79e-102.93e-11LPG ProductionPyrenesecondary5.71e-102.89e-11Natural Gas Prod.Dimethyl sulfatesecondary5.55e-102.81e-11Natural Gas Prod.1,2-Dichloroet	LPG Production	Indeno(1,2,3-cd)pyrene	secondary	9.29e-10	4.70e-11
Fuel Oil #6 Prod.2,4-Dinitrotoluenesecondary8,99e-104,56e-11Fuel Oil #4 Prod.Cobaltsecondary8,62e-104,37e-11Japanese Electric GridBenzo[a]anthracenemodel/secondary8,00e-104,05e-11Fuel Oil #2 Prod.Methyl ethyl ketonesecondary7,79e-103,95e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7,67e-103,89e-11LPG ProductionBenzo[b,j,k]fluoranthenesecondary7,07e-103,58e-11LPG ProductionMetcury compoundssecondary7,07e-103,58e-11Fuel Oil #6 Prod.Chromium (III)secondary6,96e-103,52e-11LPG ProductionBenzo[a]pyrenesecondary6,66e-103,7e-11LPG ProductionBenzo[a]pyrenesecondary6,66e-103,37e-11LPG ProductionFluorenesecondary6,66e-103,37e-11LPG ProductionFluorenesecondary6,60e-103,37e-11LPG ProductionFluorenesecondary6,60e-103,37e-11US electric gridAcetophenonemodel/secondary6,49e-103,29e-11US electric gridAcetophenonemodel/secondary5,79e-102,93e-11LPG ProductionPyrenesecondary5,71e-102,89e-11LPG ProductionPyrenesecondary5,57e-102,81e-11LPG ProductionJinethyl sulfatesecondary5,55e-102,81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5,5	Japanese Electric Grid	Ethyl Chloride	model/secondary	9.16e-10	4.64e-11
Fuel Oil #4 Prod.Cobaltsecondary8.62e-104.37e-11Japanese Electric GridBenzo[a]anthracenemodel/secondary8.00e-104.05e-11Fuel Oil #2 Prod.Methyl ethyl ketonesecondary7.79e-103.95e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.67e-103.89e-11LPG ProductionBenzo[b,j,k]fluoranthenesecondary7.36e-103.73e-11LPG ProductionMercury compoundssecondary7.07e-103.58e-11Fuel Oil #6 Prod.Chromium (III)secondary6.96e-103.52e-11LPG ProductionBenzo[a]pyrenesecondary6.66e-103.37e-11LPG ProductionBenzo[a]pyrenesecondary6.66e-103.37e-11LPG ProductionFluorenesecondary6.60e-103.32e-11LPG ProductionFluorenesecondary6.60e-103.32e-11LPG ProductionFluorenesecondary6.60e-103.32e-11US electric gridAcetophenonemodel/secondary6.04e-103.02e-11US electric gridBiphenylmodel/secondary5.79e-102.93e-11LPG ProductionPyrenesecondary5.79e-102.88e-11LPG ProductionJiptenyl sulfatesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]prylenesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]prylenesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]prylenesecondary	Fuel Oil #6 Prod.	2,4-Dinitrotoluene	secondary	8.99e-10	4.56e-11
Japanese Electric GridBenzo[a]anthracenemodel/secondary8.00e-104.05e-11Fuel Oil #2 Prod.Methyl ethyl ketonesecondary7.79e-103.95e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.67e-103.89e-11LPG ProductionBenzo[b,j,k]fluoranthenesecondary7.36e-103.73e-11LPG ProductionMercury compoundssecondary7.07e-103.58e-11Fuel Oil #6 Prod.Chromium (III)secondary6.96e-103.52e-11LPG ProductionBenzo[a]pyrenesecondary6.66e-103.37e-11LPG ProductionBenzo[a]pyrenesecondary6.66e-103.37e-11LPG ProductionFluorenesecondary6.60e-103.35e-11Fuel Oil #2 Prod.Methyl methacrylatesecondary6.60e-103.35e-11LPG ProductionFluorenesecondary6.49e-103.29e-11US electric gridAcctophenonemodel/secondary6.04e-103.06e-11US electric gridBiphenylmodel/secondary5.79e-102.93e-11LPG ProductionPyrenesecondary5.71e-102.89e-11Natural Gas Prod.Dimethyl sulfatesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Bromoformsecondary5.55e-102.81e-11	Fuel Oil #4 Prod.	Cobalt	secondary	8.62e-10	4.37e-11
Fuel Oil #2 Prod.Methyl ethyl ketonesecondary7.79e-103.95e-11Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.67e-103.89e-11LPG ProductionBenzo[b,j,k]fluoranthenesecondary7.36e-103.73e-11LPG ProductionMercury compoundssecondary7.07e-103.58e-11Fuel Oil #6 Prod.Chromium (III)secondary6.96e-103.52e-11LPG ProductionBenzo[a]pyrenesecondary6.83e-103.46e-11Fuel Oil #2 Prod.Methyl methacrylatesecondary6.66e-103.37e-11LPG ProductionFluorenesecondary6.60e-103.35e-11Fuel Oil #4 Prod.Acroleinsecondary6.04e-103.06e-11US electric gridAcetophenonemodel/secondary6.04e-103.06e-11US electric gridBiphenylmodel/secondary5.79e-102.93e-11LPG ProductionPyrenesecondary5.71e-102.89e-11Natural Gas Prod.Dimethyl sulfatesecondary5.57e-102.82e-11Fuel Oil #2 Prod.1,2-Dichloroethanesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Bromoformsecondary5.55e-102.81e-11	Japanese Electric Grid	Benzo[a]anthracene	model/secondary	8.00e-10	4.05e-11
Fuel Oil #2 Prod.Tetrachloroethylenesecondary7.67e-103.89e-11LPG ProductionBenzo[b,j,k]fluoranthenesecondary7.36e-103.73e-11LPG ProductionMercury compoundssecondary7.07e-103.58e-11Fuel Oil #6 Prod.Chromium (III)secondary6.96e-103.52e-11LPG ProductionBenzo[a]pyrenesecondary6.83e-103.46e-11Fuel Oil #2 Prod.Methyl methacrylatesecondary6.66e-103.37e-11LPG ProductionFluorenesecondary6.60e-103.35e-11LPG ProductionFluorenesecondary6.49e-103.29e-11US electric gridAcetophenonemodel/secondary6.04e-103.06e-11US electric gridBiphenylmodel/secondary5.79e-102.93e-11LPG ProductionPyrenesecondary5.57e-102.89e-11LPG ProductionDimethyl sulfatesecondary5.57e-102.82e-11Fuel Oil #2 Prod.1,2-Dichloroethanesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Benzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Benzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Bromoformsecondary5.55e-102.81e-11	Fuel Oil #2 Prod.	Methyl ethyl ketone	secondary	7.79e-10	3.95e-11
LPG ProductionBenzo[b,j,k]fluoranthenesecondary7.36e-103.73e-11LPG ProductionMercury compoundssecondary7.07e-103.58e-11Fuel Oil #6 Prod.Chromium (III)secondary6.96e-103.52e-11LPG ProductionBenzo[a]pyrenesecondary6.83e-103.46e-11Fuel Oil #2 Prod.Methyl methacrylatesecondary6.66e-103.37e-11LPG ProductionFluorenesecondary6.60e-103.35e-11LPG ProductionFluorenesecondary6.60e-103.35e-11VS electric gridAcroleinsecondary6.04e-103.06e-11US electric gridBiphenylmodel/secondary5.79e-102.93e-11LPG ProductionPyrenesecondary5.71e-102.89e-11Natural Gas Prod.Dimethyl sulfatesecondary5.57e-102.81e-11Fuel Oil #2 Prod.1,2-Dichloroethanesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Bromoformsecondary5.55e-102.81e-11	Fuel Oil #2 Prod.	Tetrachloroethylene	secondary	7.67e-10	3.89e-11
LPG ProductionMercury compoundssecondary7.07e-103.58e-11Fuel Oil #6 Prod.Chromium (III)secondary6.96e-103.52e-11LPG ProductionBenzo[a]pyrenesecondary6.83e-103.46e-11Fuel Oil #2 Prod.Methyl methacrylatesecondary6.66e-103.37e-11LPG ProductionFluorenesecondary6.60e-103.35e-11Fuel Oil #4 Prod.Acroleinsecondary6.49e-103.29e-11US electric gridAcetophenonemodel/secondary6.04e-103.06e-11US electric gridBiphenylmodel/secondary5.79e-102.93e-11LPG ProductionPyrenesecondary5.79e-102.89e-11LPG ProductionDimethyl sulfatesecondary5.57e-102.82e-11Fuel Oil #2 Prod.1,2-Dichloroethanesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Bromoformsecondary5.55e-102.81e-11	LPG Production	Benzo[b,j,k]fluoranthene	secondary	7.36e-10	3.73e-11
Fuel Oil #6 Prod.Chromium (III)secondary6.96e-103.52e-11LPG ProductionBenzo[a]pyrenesecondary6.83e-103.46e-11Fuel Oil #2 Prod.Methyl methacrylatesecondary6.66e-103.37e-11LPG ProductionFluorenesecondary6.60e-103.35e-11Fuel Oil #4 Prod.Acroleinsecondary6.49e-103.29e-11US electric gridAcetophenonemodel/secondary6.04e-103.06e-11US electric gridBiphenylmodel/secondary5.79e-102.93e-11LPG ProductionPyrenesecondary5.71e-102.89e-11LPG ProductionDimethyl sulfatesecondary5.57e-102.82e-11Fuel Oil #2 Prod.1,2-Dichloroethanesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Bromoformsecondary5.55e-102.81e-11	LPG Production	Mercury compounds	secondary	7.07e-10	3.58e-11
LPG ProductionBenzo[a]pyrenesecondary6.83e-103.46e-11Fuel Oil #2 Prod.Methyl methacrylatesecondary6.66e-103.37e-11LPG ProductionFluorenesecondary6.60e-103.35e-11Fuel Oil #4 Prod.Acroleinsecondary6.49e-103.29e-11US electric gridAcetophenonemodel/secondary6.04e-103.06e-11US electric gridBiphenylmodel/secondary5.79e-102.93e-11LPG ProductionPyrenesecondary5.71e-102.89e-11LPG ProductionDimethyl sulfatesecondary5.57e-102.82e-11Fuel Oil #2 Prod.1,2-Dichloroethanesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Bromoformsecondary5.55e-102.81e-11	Fuel Oil #6 Prod.	Chromium (III)	secondary	6.96e-10	3.52e-11
Fuel Oil #2 Prod.Methyl methacrylatesecondary6.66e-103.37e-11LPG ProductionFluorenesecondary6.60e-103.35e-11Fuel Oil #4 Prod.Acroleinsecondary6.49e-103.29e-11US electric gridAcetophenonemodel/secondary6.04e-103.06e-11US electric gridBiphenylmodel/secondary5.79e-102.93e-11LPG ProductionPyrenesecondary5.71e-102.89e-11Natural Gas Prod.Dimethyl sulfatesecondary5.57e-102.82e-11Fuel Oil #2 Prod.1,2-Dichloroethanesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Bromoformsecondary5.44e-102.76e-11	LPG Production	Benzo[a]pyrene	secondary	6.83e-10	3.46e-11
LPG ProductionFluorenesecondary6.60e-103.35e-11Fuel Oil #4 Prod.Acroleinsecondary6.49e-103.29e-11US electric gridAcetophenonemodel/secondary6.04e-103.06e-11US electric gridBiphenylmodel/secondary5.79e-102.93e-11LPG ProductionPyrenesecondary5.71e-102.89e-11Natural Gas Prod.Dimethyl sulfatesecondary5.57e-102.82e-11Fuel Oil #2 Prod.1,2-Dichloroethanesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Bromoformsecondary5.44e-102.76e-11	Fuel Oil #2 Prod.	Methyl methacrylate	secondary	6.66e-10	3.37e-11
Fuel Oil #4 Prod.Acroleinsecondary6.49e-103.29e-11US electric gridAcetophenonemodel/secondary6.04e-103.06e-11US electric gridBiphenylmodel/secondary5.79e-102.93e-11LPG ProductionPyrenesecondary5.71e-102.89e-11Natural Gas Prod.Dimethyl sulfatesecondary5.57e-102.82e-11Fuel Oil #2 Prod.1,2-Dichloroethanesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Bromformsecondary5.44e-102.76e-11	LPG Production	Fluorene	secondary	6.60e-10	3.35e-11
US electric gridAcetophenonemodel/secondary6.04e-103.06e-11US electric gridBiphenylmodel/secondary5.79e-102.93e-11LPG ProductionPyrenesecondary5.71e-102.89e-11Natural Gas Prod.Dimethyl sulfatesecondary5.57e-102.82e-11Fuel Oil #2 Prod.1,2-Dichloroethanesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Bromoformsecondary5.44e-102.76e-11	Fuel Oil #4 Prod.	Acrolein	secondary	6.49e-10	3.29e-11
US electric gridBiphenylmodel/secondary5.79e-102.93e-11LPG ProductionPyrenesecondary5.71e-102.89e-11Natural Gas Prod.Dimethyl sulfatesecondary5.57e-102.82e-11Fuel Oil #2 Prod.1,2-Dichloroethanesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Bromoformsecondary5.44e-102.76e-11	US electric grid	Acetophenone	model/secondary	6.04e-10	3.06e-11
LPG ProductionPyrenesecondary5.71e-102.89e-11Natural Gas Prod.Dimethyl sulfatesecondary5.57e-102.82e-11Fuel Oil #2 Prod.1,2-Dichloroethanesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Bromoformsecondary5.44e-102.76e-11	US electric grid	Biphenyl	model/secondary	5.79e-10	2.93e-11
Natural Gas Prod.Dimethyl sulfatesecondary5.57e-102.82e-11Fuel Oil #2 Prod.1,2-Dichloroethanesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Bromoformsecondary5.44e-102.76e-11	LPG Production	Pyrene	secondary	5.71e-10	2.89e-11
Fuel Oil #2 Prod.1,2-Dichloroethanesecondary5.55e-102.81e-11LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Bromoformsecondary5.44e-102.76e-11	Natural Gas Prod.	Dimethyl sulfate	secondary	5.57e-10	2.82e-11
LPG ProductionBenzo[g,h,i]perylenesecondary5.55e-102.81e-11Fuel Oil #2 Prod.Bromoformsecondary5.44e-102.76e-11	Fuel Oil #2 Prod.	1,2-Dichloroethane	secondary	5.55e-10	2.81e-11
Fuel Oil #2 Prod.Bromoformsecondary5.44e-102.76e-11	LPG Production	Benzo[g,h,i]perylene	secondary	5.55e-10	2.81e-11
	Fuel Oil #2 Prod.	Bromoform	secondary	5.44e-10	2.76e-11

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
LPG Production	Fluoranthene	secondary	5.39e-10	2.73e-11
Japanese Electric Grid	Chrysene	model/secondary	5.35e-10	2.71e-11
Natural Gas Prod.	Isophorone	secondary	5.34e-10	2.71e-11
Fuel Oil #4 Prod.	Methyl chloride	secondary	5.26e-10	2.66e-11
Fuel Oil #2 Prod.	Dichloromethane	secondary	5.25e-10	2.66e-11
Natural Gas Prod.	Zinc (+2)	secondary	5.24e-10	2.66e-11
LPG Production	HALON-1301	secondary	5.08e-10	2.57e-11
LPG Production	Benzo[b]fluoranthene	secondary	4.99e-10	2.53e-11
Japanese Electric Grid	Acetophenone	model/secondary	4.82e-10	2.44e-11
Fuel Oil #6 Prod.	Aromatic hydrocarbons	secondary	4.74e-10	2.40e-11
Natural Gas Prod.	Chlorine	secondary	4.73e-10	2.40e-11
Fuel Oil #4 Prod.	Beryllium	secondary	4.67e-10	2.36e-11
Glass/frit	Chromium	primary	4.67e-10	2.36e-11
Glass/frit	Manganese	primary	4.67e-10	2.36e-11
Japanese Electric Grid	Biphenyl	model/secondary	4.62e-10	2.34e-11
Japanese Electric Grid	Indeno(1,2,3-cd)pyrene	model/secondary	4.48e-10	2.27e-11
Fuel Oil #6 Prod.	Ethylbenzene	secondary	4.47e-10	2.26e-11
Fuel Oil #6 Prod.	Copper (+1 & +2)	secondary	4.43e-10	2.25e-11
Fuel Oil #2 Prod.	Chlorobenzene	secondary	4.40e-10	2.23e-11
Natural Gas Prod.	Methyl ethyl ketone	secondary	4.31e-10	2.18e-11
Japanese Electric Grid	Benzo[g,h,i]perylene	model/secondary	4.30e-10	2.18e-11
Fuel Oil #6 Prod.	o-xylene	secondary	4.29e-10	2.17e-11
Natural Gas Prod.	Tetrachloroethylene	secondary	4.24e-10	2.15e-11
Fuel Oil #2 Prod.	Aromatic hydrocarbons	secondary	3.93e-10	1.99e-11
Japanese Electric Grid	Benzo[b,j,k]fluoranthene	model/secondary	3.89e-10	1.97e-11
Fuel Oil #4 Prod.	Methyl hydrazine	secondary	3.81e-10	1.93e-11
Natural Gas Prod.	Silicon	secondary	3.71e-10	1.88e-11
Natural Gas Prod.	Methyl methacrylate	secondary	3.68e-10	1.87e-11
Fuel Oil #2 Prod.	Copper (+1 & +2)	secondary	3.68e-10	1.86e-11
US electric grid	Acenaphthylene	model/secondary	3.60e-10	1.82e-11
LPG Production	Dibenzo[a,h]anthracene	secondary	3.55e-10	1.80e-11
Fuel Oil #2 Prod.	2,4-Dinitrotoluene	secondary	3.50e-10	1.77e-11
Japanese Electric Grid	Acenaphthylene	model/secondary	3.29e-10	1.67e-11
Natural Gas Prod.	Barium cmpds	secondary	3.21e-10	1.63e-11
Glass/frit	Lead	primary	3.20e-10	1.62e-11
US electric grid	Acenaphthene	model/secondary	3.08e-10	1.56e-11
Natural Gas Prod.	1,2-Dichloroethane	secondary	3.07e-10	1.55e-11
Natural Gas Prod.	Bromoform	secondary	3.01e-10	1.52e-11
Japanese Electric Grid	Dibenzo[a,h]anthracene	model/secondary	2.96e-10	1.50e-11
Fuel Oil #4 Prod.	Acetaldehyde	secondary	2.92e-10	1.48e-11
Fuel Oil #4 Prod.	Propionaldehyde	secondary	2.92e-10	1.48e-11
Natural Gas Prod.	Dichloromethane	secondary	2.90e-10	1.47e-11
Fuel Oil #4 Prod.	Mercury	secondary	2.89e-10	1.46e-11
Fuel Oil #2 Prod.	Chromium (III)	secondary	2.70e-10	1.37e-11
Fuel Oil #4 Prod.	Cadmium	secondary	2.67e-10	1.35e-11
Natural Gas Prod.	Chlorobenzene	secondary	2.43e-10	1.23e-11
Japanese Electric Grid	2-Methylnaphthalene	model/secondary	2.38e-10	1.20e-11
Fuel Oil #6 Prod.	Methyl tert-butyl ether	secondary	2.25e-10	1.14e-11
Fuel Oil #4 Prod.	Di(2-ethylhexyl)phthalate	secondary	2.25e-10	1.14e-11
Fuel Oil #2 Prod.	o-xylene	secondary	2.06e-10	1.05e-11

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Japanese Electric Grid	Pyrene	model/secondary	2.00e-10	1.01e-11
Natural Gas Prod.	2,4-Dinitrotoluene	secondary	1.93e-10	9.80e-12
Fuel Oil #4 Prod.	Toluene	secondary	1.90e-10	9.60e-12
Fuel Oil #2 Prod.	Ethylbenzene	secondary	1.74e-10	8.83e-12
Japanese Electric Grid	Fluorene	model/secondary	1.74e-10	8.82e-12
Fuel Oil #6 Prod.	Phenanthrene	secondary	1.71e-10	8.68e-12
US electric grid	Benzo[b,j,k]fluoranthene	model/secondary	1.70e-10	8.60e-12
Natural Gas Prod.	3-Methylcholanthrene	secondary	1.69e-10	8.55e-12
Japanese Electric Grid	Fluoranthene	model/secondary	1.66e-10	8.43e-12
Fuel Oil #6 Prod.	Styrene	secondary	1.64e-10	8.31e-12
US electric grid	Chrysene	model/secondary	1.63e-10	8.26e-12
Natural Gas Prod.	Chromium (III)	secondary	1.61e-10	8.16e-12
Fuel Oil #6 Prod.	Vinyl acetate	secondary	1.60e-10	8.11e-12
Fuel Oil #4 Prod.	Cadmium empds	secondary	1.58e-10	8.02e-12
US electric grid	Benzo[a]anthracene	model/secondary	1.48e-10	7.50e-12
LPG Production	5-Methyl chrysene	secondary	1.47e-10	7.46e-12
Glass/frit	Cobalt	primary	1.43e-10	7.26e-12
US electric grid	Fluorene	model/secondary	1.28e-10	6.46e-12
LPG Production	Halogenated matter (organic)	secondary	1.17e-10	5.91e-12
Fuel Oil #4 Prod.	Dimethyl sulfate	secondary	1.07e-10	5.44e-12
Fuel Oil #6 Prod.	3-Methylcholanthrene	secondary	1.07e-10	5.43e-12
US electric grid	Indeno(1,2,3-cd)pyrene	model/secondary	1.05e-10	5.33e-12
Fuel Oil #4 Prod.	Isophorone	secondary	1.03e-10	5.22e-12
US electric grid	Fluoranthene	model/secondary	1.02e-10	5.17e-12
Fuel Oil #4 Prod.	1,4-Dichlorobenzene	secondary	1.00e-10	5.09e-12
Natural Gas Prod.	Ethylbenzene	secondary	9.66e-11	4.89e-12
Fuel Oil #2 Prod.	Methyl tert-butyl ether	secondary	8.74e-11	4.43e-12
Natural Gas Prod.	Aluminum (elemental)	secondary	8.50e-11	4.30e-12
US electric grid	Pyrene	model/secondary	8.43e-11	4.27e-12
Fuel Oil #4 Prod.	Methyl ethyl ketone	secondary	8.31e-11	4.21e-12
Fuel Oil #4 Prod.	Tetrachloroethylene	secondary	8.18e-11	4.14e-12
Fuel Oil #4 Prod.	Methyl methacrylate	secondary	7.10e-11	3.60e-12
Natural Gas Prod.	Phenanthrene	secondary	7.05e-11	3.57e-12
Fuel Oil #2 Prod.	Phenanthrene	secondary	7.03e-11	3.56e-12
Fuel Oil #6 Prod.	Ethylene dibromide	secondary	6.48e-11	3.28e-12
Fuel Oil #2 Prod.	Styrene	secondary	6.38e-11	3.23e-12
Fuel Oil #2 Prod.	Vinvl acetate	secondary	6.23e-11	3.15e-12
Fuel Oil #6 Prod.	1.1.1-Trichloroethane	secondary	6.11e-11	3.09e-12
US electric grid	o-xvlene	model/secondary	6.06e-11	3.07e-12
Fuel Oil #4 Prod.	1.2-Dichloroethane	secondary	5.92e-11	3.00e-12
Fuel Oil #4 Prod	Bromoform	secondary	5 80e-11	2.94e-12
US electric grid	Benzolg.h.ilpervlene	model/secondary	5.75e-11	2.91e-12
Fuel Oil #4 Prod.	Dichloromethane	secondary	5.60e-11	2.84e-12
Fuel Oil #2 Prod	3-Methylcholanthrene	secondary	5 57e-11	2.82e-12
US electric grid	Benzo[a]pyrene	model/secondary	5.44e-11	2.76e-12
Natural Gas Prod.	2-Methylnaphthalene	secondary	5.41e-11	2.74e-12
Fuel Oil #6 Prod.	$Z_{\text{inc}}(+2)$	secondary	5 33e-11	2.70e-12
Natural Gas Prod.	Methyl tert-butyl ether	secondary	4 83e-11	2.45e-12
Fuel Oil #4 Prod	Chlorobenzene	secondary	4 69e-11	2.37e-12
Japanese Electric Grid	Benzo[a]pyrene	model/secondary	4 35e-11	2.370 12 2.20e-12
apartose Electric Orla	2 on Lola la l	inouch secondary	1.550-11	2.200 12

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Fuel Oil #6 Prod.	Ethyl Chloride	secondary	4.33e-11	2.19e-12
US electric grid	2-Methylnaphthalene	model/secondary	4.26e-11	2.16e-12
Fuel Oil #4 Prod.	2,4-Dinitrotoluene	secondary	3.73e-11	1.89e-12
Fuel Oil #6 Prod.	Cumene	secondary	3.66e-11	1.85e-12
Natural Gas Prod.	Styrene	secondary	3.53e-11	1.79e-12
Natural Gas Prod.	Vinyl acetate	secondary	3.44e-11	1.74e-12
Fuel Oil #6 Prod.	2-Methylnaphthalene	secondary	3.43e-11	1.74e-12
Fuel Oil #2 Prod.	Zinc (+2)	secondary	3.39e-11	1.72e-12
Fuel Oil #4 Prod.	Aromatic hydrocarbons	secondary	3.19e-11	1.62e-12
US electric grid	5-Methyl chrysene	model/secondary	3.15e-11	1.60e-12
Fuel Oil #4 Prod.	Copper (+1 & +2)	secondary	2.99e-11	1.51e-12
Fuel Oil #6 Prod.	Chlorine	secondary	2.96e-11	1.50e-12
Fuel Oil #4 Prod.	Chromium (III)	secondary	2.88e-11	1.46e-12
Fuel Oil #2 Prod.	Ethylene dibromide	secondary	2.52e-11	1.28e-12
Japanese Electric Grid	5-Methyl chrysene	model/secondary	2.51e-11	1.27e-12
LPG Production	Anthracene	secondary	2.48e-11	1.26e-12
Fuel Oil #2 Prod.	1,1,1-Trichloroethane	secondary	2.37e-11	1.20e-12
Fuel Oil #6 Prod.	Acetophenone	secondary	2.28e-11	1.15e-12
Fuel Oil #6 Prod.	Biphenyl	secondary	2.18e-11	1.11e-12
Fuel Oil #4 Prod.	o-xylene	secondary	2.01e-11	1.02e-12
Fuel Oil #4 Prod.	Ethylbenzene	secondary	1.86e-11	9.40e-13
Fuel Oil #2 Prod.	2-Methylnaphthalene	secondary	1.78e-11	9.04e-13
Fuel Oil #2 Prod.	Ethyl Chloride	secondary	1.68e-11	8.52e-13
Fuel Oil #6 Prod.	Acenaphthylene	secondary	1.61e-11	8.17e-13
Fuel Oil #2 Prod.	Chlorine	secondary	1.56e-11	7.90e-13
Fuel Oil #2 Prod.	Cumene	secondary	1.42e-11	7.21e-13
Natural Gas Prod.	Cadmium cmpds	secondary	1.40e-11	7.11e-13
US electric grid	Dibenzo[a,h]anthracene	model/secondary	1.40e-11	7.07e-13
Natural Gas Prod.	Ethylene dibromide	secondary	1.39e-11	7.06e-13
Natural Gas Prod.	1,1,1-Trichloroethane	secondary	1.31e-11	6.65e-13
Fuel Oil #6 Prod.	Acenaphthene	secondary	1.19e-11	6.01e-13
Fuel Oil #4 Prod.	Methyl tert-butyl ether	secondary	9.32e-12	4.72e-13
Natural Gas Prod.	Ethyl Chloride	secondary	9.30e-12	4.71e-13
Fuel Oil #2 Prod.	Acetophenone	secondary	8.86e-12	4.49e-13
Fuel Oil #2 Prod.	Biphenyl	secondary	8.49e-12	4.30e-13
Fuel Oil #6 Prod.	Chrysene	secondary	8.24e-12	4.18e-13
Natural Gas Prod.	Cumene	secondary	7.87e-12	3.99e-13
Fuel Oil #6 Prod.	Benzo[a]anthracene	secondary	7.80e-12	3.95e-13
Fuel Oil #4 Prod.	Phenanthrene	secondary	7.32e-12	3.71e-13
Natural Gas Prod.	Acenaphthylene	secondary	7.00e-12	3.54e-13
Fuel Oil #4 Prod.	Styrene	secondary	6.80e-12	3.45e-13
Fuel Oil #4 Prod.	Vinyl acetate	secondary	6.64e-12	3.36e-13
Fuel Oil #2 Prod.	Acenaphthylene	secondary	6.63e-12	3.36e-13
Fuel Oil #6 Prod.	Indeno(1,2,3-cd)pyrene	secondary	6.11e-12	3.09e-13
Fuel Oil #6 Prod.	Benzo[b,j,k]fluoranthene	secondary	5.94e-12	3.01e-13
Natural Gas Prod.	Benzo[a]anthracene	secondary	5.61e-12	2.84e-13
Japanese Electric Grid	Anthracene	model/secondarv	5.42e-12	2.75e-13
Fuel Oil #2 Prod.	Acenaphthene	secondary	5.41e-12	2.74e-13
Natural Gas Prod.	Chrysene	secondary	5.40e-12	2.74e-13
Fuel Oil #4 Prod.	3-Methylcholanthrene	secondary	5.27e-12	2.67e-13
		-		

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Fuel Oil #6 Prod.	Fluorene	secondary	5.11e-12	2.59e-13
Fuel Oil #6 Prod.	Benzo[a]pyrene	secondary	5.10e-12	2.58e-13
Natural Gas Prod.	Indeno(1,2,3-cd)pyrene	secondary	4.93e-12	2.50e-13
Natural Gas Prod.	Acetophenone	secondary	4.90e-12	2.48e-13
Fuel Oil #6 Prod.	Nickel cmpds	secondary	4.70e-12	2.38e-13
Natural Gas Prod.	Biphenyl	secondary	4.70e-12	2.38e-13
Natural Gas Prod.	Acenaphthene	secondary	4.50e-12	2.28e-13
Natural Gas Prod.	Benzo[b]fluoranthene	secondary	4.17e-12	2.11e-13
Fuel Oil #6 Prod.	Fluoranthene	secondary	4.11e-12	2.08e-13
Fuel Oil #6 Prod.	Pyrene	secondary	4.04e-12	2.05e-13
Fuel Oil #6 Prod.	Lead cmpds	secondary	3.95e-12	2.00e-13
Fuel Oil #2 Prod.	Nickel cmpds	secondary	3.90e-12	1.97e-13
Fuel Oil #2 Prod.	Benzo[a]anthracene	secondary	3.76e-12	1.90e-13
US electric grid	Anthracene	model/secondary	3.70e-12	1.87e-13
Fuel Oil #2 Prod.	Chrysene	secondary	3.66e-12	1.85e-13
Fuel Oil #6 Prod.	Benzo[g,h,i]perylene	secondary	3.43e-12	1.74e-13
Fuel Oil #2 Prod.	Lead cmpds	secondary	3.27e-12	1.66e-13
Natural Gas Prod.	Benzo[g,h,i]perylene	secondary	3.19e-12	1.62e-13
Natural Gas Prod.	Benzo[a]pyrene	secondary	3.15e-12	1.59e-13
Fuel Oil #4 Prod.	Zinc (+2)	secondary	2.98e-12	1.51e-13
Natural Gas Prod.	Dibenzo[a,h]anthracene	secondary	2.83e-12	1.44e-13
Natural Gas Prod.	Aromatic hydrocarbons	secondary	2.83e-12	1.43e-13
Fuel Oil #2 Prod.	Indeno(1,2,3-cd)pyrene	secondary	2.82e-12	1.43e-13
Fuel Oil #6 Prod.	Benzo[b]fluoranthene	secondary	2.74e-12	1.39e-13
Fuel Oil #4 Prod.	Ethylene dibromide	secondary	2.69e-12	1.36e-13
Natural Gas Prod.	Copper (+1 & +2)	secondary	2.65e-12	1.34e-13
Fuel Oil #4 Prod.	1,1,1-Trichloroethane	secondary	2.53e-12	1.28e-13
Natural Gas Prod.	Pyrene	secondary	2.45e-12	1.24e-13
Fuel Oil #6 Prod.	Mercury compounds	secondary	2.39e-12	1.21e-13
Fuel Oil #2 Prod.	Benzo[b,j,k]fluoranthene	secondary	2.31e-12	1.17e-13
Fuel Oil #2 Prod.	Benzo[a]pyrene	secondary	2.11e-12	1.07e-13
Fuel Oil #2 Prod.	Fluorene	secondary	2.06e-12	1.04e-13
Fuel Oil #2 Prod.	Mercury compounds	secondary	1.98e-12	1.00e-13
Fuel Oil #6 Prod.	Dibenzo[a,h]anthracene	secondary	1.90e-12	9.65e-14
Fuel Oil #4 Prod.	Ethyl Chloride	secondary	1.79e-12	9.09e-14
Fuel Oil #2 Prod.	Pyrene	secondary	1.75e-12	8.87e-14
Fuel Oil #6 Prod.	HALON-1301	secondary	1.72e-12	8.70e-14
Fuel Oil #4 Prod.	2-Methylnaphthalene	secondary	1.69e-12	8.55e-14
Fuel Oil #2 Prod.	Fluoranthene	secondary	1.67e-12	8.48e-14
Fuel Oil #2 Prod.	Benzo[g,h,i]perylene	secondary	1.67e-12	8.44e-14
Natural Gas Prod.	Fluorene	secondary	1.64e-12	8.30e-14
Fuel Oil #4 Prod.	Cumene	secondary	1.52e-12	7.69e-14
Fuel Oil #2 Prod.	Benzo[b]fluoranthene	secondary	1.47e-12	7.47e-14
Fuel Oil #4 Prod.	Chlorine	secondary	1.47e-12	7.44e-14
Natural Gas Prod.	Fluoranthene	secondary	1.46e-12	7.42e-14
Fuel Oil #2 Prod.	HALON-1301	secondary	1.42e-12	7.22e-14
Natural Gas Prod.	Benzo[b,j,k]fluoranthene	secondary	1.28e-12	6.47e-14
Fuel Oil #6 Prod.	5-Methyl chrysene	secondary	1.19e-12	6.02e-14
Fuel Oil #2 Prod.	Dibenzo[a,h]anthracene	secondary	1.05e-12	5.30e-14
Fuel Oil #4 Prod.	Acetophenone	secondary	9.45e-13	4.78e-14

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Fuel Oil #4 Prod.	Biphenyl	secondary	9.06e-13	4.59e-14
Fuel Oil #4 Prod.	Acenaphthylene	secondary	6.90e-13	3.50e-14
Fuel Oil #4 Prod.	Acenaphthene	secondary	5.39e-13	2.73e-14
Fuel Oil #2 Prod.	5-Methyl chrysene	secondary	4.62e-13	2.34e-14
Fuel Oil #6 Prod.	Halogenated matter (organic)	secondary	3.95e-13	2.00e-14
Fuel Oil #4 Prod.	Chrysene	secondary	3.68e-13	1.87e-14
Fuel Oil #4 Prod.	Benzo[a]anthracene	secondary	3.66e-13	1.85e-14
Fuel Oil #2 Prod.	Halogenated matter (organic)	secondary	3.27e-13	1.66e-14
Fuel Oil #4 Prod.	Nickel cmpds	secondary	3.17e-13	1.60e-14
LPG Production	Halogenated hydrocarbons (unspecified)	secondary	2.92e-13	1.48e-14
Fuel Oil #4 Prod.	Indeno(1,2,3-cd)pyrene	secondary	2.79e-13	1.41e-14
Fuel Oil #4 Prod.	Lead cmpds	secondary	2.66e-13	1.35e-14
Natural Gas Prod.	5-Methyl chrysene	secondary	2.55e-13	1.29e-14
Fuel Oil #4 Prod.	Benzo[b,j,k]fluoranthene	secondary	2.46e-13	1.25e-14
Fuel Oil #4 Prod.	Benzo[a]pyrene	secondary	2.19e-13	1.11e-14
Fuel Oil #4 Prod.	Fluorene	secondary	2.16e-13	1.09e-14
Fuel Oil #6 Prod.	Anthracene	secondary	1.79e-13	9.07e-15
Fuel Oil #4 Prod.	Pyrene	secondary	1.78e-13	9.02e-15
Fuel Oil #4 Prod	Fluoranthene	secondary	1 75e-13	8 86e-15
Fuel Oil #4 Prod	Benzolg h ilpervlene	secondary	1 62e-13	8 20e-15
Fuel Oil #4 Prod	Mercury compounds	secondary	1.61e-13	8 16e-15
Fuel Oil #4 Prod	Benzo[b]fluoranthene	secondary	1 38e-13	6.98e-15
Fuel Oil #4 Prod	HALON-1301	secondary	1.56e-13	5.86e-15
Fuel Oil #4 Prod	Dibenzo[a h]anthracene	secondary	9 69e-14	4 91e-15
Natural Gas Prod	Anthracene	secondary	9.56e-14	4 84e-15
Fuel Oil #2 Prod	Anthracene	secondary	7.64e-14	3.87e-15
US electric grid	2 3 7 8-TCDF	model/secondary	7.04e 14	3.70e-15
Japanese Electric Grid	2,3,7,8 TODE	model/secondary	5.84e-14	2.96e-15
Fuel Oil #4 Prod	5-Methyl chrysene	secondary	4 92e-14	2.90e-15
Natural Gas Prod	Nickel cmpds	secondary	7.520-14 2.81e-14	1.42e-15
Fuel Oil #4 Prod	Halogenated matter (organic)	secondary	2.610-14	1.35e 15
Natural Gas Prod	Lead cmpds	secondary	2.000-14	1.100.15
Natural Gas Prod	Mercury compounds	secondary	1 /30 1/	7.240.16
Natural Gas Prod	HALON 1201	secondary	1.430-14	5 202 16
Fuel Oil #4 Prod	Anthracene	secondary	7.820.15	3.206-10
Natural Gas Brod	Halogeneted motter (organia)	secondary	7.826-13	1 100 16
Final Oil #6 Drod	Halogenated matter (organic)	secondary	2.306-13	5.002.17
Fuel Oil #0 Prod.	Halogenated hydrocarbons (unspecified)	secondary	9.8/6-10	3.000-17
Fuel Oil #2 Prod.	Halogenated hydrocarbons (unspecified)	secondary	8.196-10	4.15e-17
Fuel OII #4 Prod.	Halogenated hydrocarbons (unspecified)	secondary	6.05e-17	3.376-18
Natural Gas Prod.	Halogenated hydrocarbons (unspecified)	secondary	5.896-18	2.98e-19
Total Manufacturing			9.46e+01	4.79e+00
Use, Maintenance and Repair Life	-cycle Stage			
US electric grid	Sulfur dioxide	model/secondary	1.65e+03	8.34e+01
US electric grid	Nitrogen oxides	model/secondary	1.18e+00	5.96e-02
US electric grid	Carbon monoxide	model/secondary	1.01e+00	5.12e-02
US electric grid	Methane	model/secondary	6.45e-01	3.27e-02
US electric grid	Arsenic	model/secondary	5.58e-01	2.83e-02
US electric grid	Hydrochloric acid	model/secondary	4.93e-01	2.50e-02
US electric grid	Selenium	model/secondary	9.28e-02	4.70e-03

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
US electric grid	Vanadium	model/secondary	6.98e-02	3.54e-03
US electric grid	PM-10	model/secondary	5.78e-02	2.93e-03
US electric grid	Hydrofluoric acid	model/secondary	1.35e-02	6.81e-04
US electric grid	Formaldehyde	model/secondary	9.69e-03	4.91e-04
US electric grid	Benzene	model/secondary	6.97e-03	3.53e-04
US electric grid	Phosphorus (yellow or white)	model/secondary	3.93e-03	1.99e-04
US electric grid	Nitrous oxide	model/secondary	3.40e-03	1.72e-04
US electric grid	Zinc (elemental)	model/secondary	1.81e-03	9.18e-05
US electric grid	Antimony	model/secondary	1.48e-03	7.51e-05
US electric grid	Molybdenum	model/secondary	4.96e-04	2.51e-05
US electric grid	2-Chloroacetophenone	model/secondary	4.31e-04	2.18e-05
US electric grid	Bromomethane	model/secondary	4.27e-04	2.16e-05
US electric grid	Cyanide (-1)	model/secondary	2.24e-04	1.14e-05
US electric grid	2,3,7,8-TCDD	model/secondary	1.70e-04	8.59e-06
US electric grid	Nickel	model/secondary	1.63e-04	8.26e-06
US electric grid	Naphthalene	model/secondary	1.35e-04	6.84e-06
US electric grid	Barium	model/secondary	1.16e-04	5.90e-06
US electric grid	Carbon disulfide	model/secondary	8.01e-05	4.06e-06
US electric grid	Benzyl chloride	model/secondary	6.28e-05	3.18e-06
US electric grid	Chloroform	model/secondary	4.88e-05	2.47e-06
US electric grid	Manganese	model/secondary	4.56e-05	2.31e-06
US electric grid	Chromium (VI)	model/secondary	3.43e-05	1.74e-06
US electric grid	Acrolein	model/secondary	2.60e-05	1.32e-06
US electric grid	Copper	model/secondary	2.24e-05	1.13e-06
US electric grid	Methyl chloride	model/secondary	2.11e-05	1.07e-06
US electric grid	Fluoride	model/secondary	1.95e-05	9.89e-07
US electric grid	Methyl hydrazine	model/secondary	1.52e-05	7.72e-07
US electric grid	Lead	model/secondary	1.27e-05	6.46e-07
US electric grid	Cobalt	model/secondary	1.22e-05	6.16e-07
US electric grid	Propionaldehyde	model/secondary	1.17e-05	5.93e-07
US electric grid	Acetaldehyde	model/secondary	1.17e-05	5.93e-07
US electric grid	Di(2-ethylhexyl)phthalate	model/secondary	8.99e-06	4.56e-07
US electric grid	Mercury	model/secondary	7.51e-06	3.81e-07
US electric grid	Hexane	model/secondary	6.01e-06	3.04e-07
US electric grid	Cadmium	model/secondary	4.82e-06	2.44e-07
US electric grid	Dimethyl sulfate	model/secondary	4.30e-06	2.18e-07
US electric grid	Toluene	model/secondary	4.25e-06	2.15e-07
US electric grid	Isophorone	model/secondary	4.13e-06	2.09e-07
US electric grid	Methyl ethyl ketone	model/secondary	3.33e-06	1.69e-07
US electric grid	Tetrachloroethylene	model/secondary	3.28e-06	1.66e-07
US electric grid	Methyl methacrylate	model/secondary	2.85e-06	1.44e-07
US electric grid	1,2-Dichloroethane	model/secondary	2.37e-06	1.20e-07
US electric grid	Bromoform	model/secondary	2.33e-06	1.18e-07
US electric grid	Dichloromethane	model/secondary	2.24e-06	1.14e-07
US electric grid	Beryllium	model/secondary	1.91e-06	9.67e-08
US electric grid	Chlorobenzene	model/secondary	1.88e-06	9.52e-08
US electric grid	2,4-Dinitrotoluene	model/secondary	1.49e-06	7.57e-08
US electric grid	Ethylbenzene	model/secondary	7.41e-07	3.75e-08
US electric grid	Methyl tert-butyl ether	model/secondary	3.74e-07	1.89e-08
US electric grid	Phenol	model/secondary	2.85e-07	1.44e-08

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
US electric grid	Styrene	model/secondary	2.73e-07	1.38e-08
US electric grid	Vinyl acetate	model/secondary	2.66e-07	1.35e-08
US electric grid	Phenanthrene	model/secondary	2.51e-07	1.27e-08
US electric grid	Xylene (mixed isomers)	model/secondary	2.21e-07	1.12e-08
US electric grid	Chromium (III)	model/secondary	1.97e-07	9.98e-09
US electric grid	1,1,1-Trichloroethane	model/secondary	1.08e-07	5.49e-09
US electric grid	Ethylene dibromide	model/secondary	1.08e-07	5.45e-09
US electric grid	Ethyl Chloride	model/secondary	7.19e-08	3.64e-09
US electric grid	Cumene	model/secondary	6.08e-08	3.08e-09
US electric grid	Acetophenone	model/secondary	3.78e-08	1.92e-09
US electric grid	Biphenyl	model/secondary	3.63e-08	1.84e-09
US electric grid	Acenaphthylene	model/secondary	2.26e-08	1.14e-09
US electric grid	Acenaphthene	model/secondary	1.93e-08	9.78e-10
US electric grid	Benzo[b,j,k]fluoranthene	model/secondary	1.06e-08	5.39e-10
US electric grid	Chrysene	model/secondary	1.02e-08	5.17e-10
US electric grid	Benzo[a]anthracene	model/secondary	9.27e-09	4.70e-10
US electric grid	Fluorene	model/secondary	7.99e-09	4.05e-10
US electric grid	Indeno(1,2,3-cd)pyrene	model/secondary	6.59e-09	3.34e-10
US electric grid	Fluoranthene	model/secondary	6.39e-09	3.24e-10
US electric grid	Pyrene	model/secondary	5.28e-09	2.68e-10
US electric grid	o-xylene	model/secondary	3.79e-09	1.92e-10
US electric grid	Benzo[g,h,i]perylene	model/secondary	3.60e-09	1.83e-10
US electric grid	Benzo[a]pyrene	model/secondary	3.41e-09	1.73e-10
US electric grid	2-Methylnaphthalene	model/secondary	2.67e-09	1.35e-10
US electric grid	5-Methyl chrysene	model/secondary	1.97e-09	1.00e-10
US electric grid	Dibenzo[a,h]anthracene	model/secondary	8.74e-10	4.43e-11
US electric grid	Anthracene	model/secondary	2.32e-10	1.17e-11
US electric grid	2,3,7,8-TCDF	model/secondary	4.57e-12	2.32e-13
Total Use, Maintenance and Repia	ır		1.65e+03	8.36e+01
End-of-life Life-cycle Stage				
CRT Incineration	Sulfur dioxide	secondary	1.98e-01	1.00e-02
CRT landfilling	Sulfur dioxide	primary	1.85e-01	9.37e-03
US electric grid	Sulfur dioxide	model/secondary	1.65e-01	8.35e-03
CRT landfilling	Carbon monoxide	primary	9.22e-03	4.67e-04
CRT landfilling	Nitrogen dioxide	primary	1.85e-03	9.38e-05
CRT landfilling	PM	primary	2.08e-04	1.06e-05
US electric grid	Nitrogen oxides	model/secondary	1.18e-04	5.97e-06
CRT landfilling	Arsenic empds	primary	1.02e-04	5.18e-06
US electric grid	Carbon monoxide	model/secondary	1.01e-04	5.12e-06
CRT Incineration	Arsenic empds	secondary	9.83e-05	4.98e-06
US electric grid	Methane	model/secondary	6.45e-05	3.27e-06
US electric grid	Arsenic	model/secondary	5.58e-05	2.83e-06
CRT landfilling	Methane	primary	5.44e-05	2.76e-06
US electric grid	Hydrochloric acid	model/secondary	4.93e-05	2.50e-06
LPG Production	Carbon monoxide	secondary	4.61e-05	2.34e-06
CRT landfilling	Benzene	primary	4.04e-05	2.05e-06
CRT Incineration	Lead	secondary	1.43e-05	7.24e-07
CRT Incineration	Barium cmpds	secondary	1.27e-05	6.45e-07
CRT Incineration	Silver compounds	secondary	1.12e-05	5.68e-07
CRT landfilling	Silver compounds	primary	1.05e-05	5.30e-07

CRT landfillingBarium empdsprimary1.01e-055.13e-07US electric gridSeleniummodel/secondary9.28e-064.70e-07LPG ProductionVanadiumsecondary9.04e-064.58e-07LPG ProductionBenzenesecondary7.40e-063.75e-07LPG ProductionMethanesecondary7.24e-063.67e-07US electric gridVanadiummodel/secondary6.98e-063.54e-07LPG ProductionSulfur oxidessecondary6.92e-063.50e-07US electric gridPM-10model/secondary5.78e-062.93e-07LPG ProductionNitrogen oxidessecondary4.94e-062.51e-07CRT landfillingHydrochloric acidprimary4.69e-062.37e-07CRT landfillingAmmoniaprimary1.16e-065.86e-08US electric gridHydrofluoric acidmodel/secondary1.35e-066.82e-08LPG ProductionFormaldehydesecondary1.16e-065.86e-08LPG ProductionFormaldehydesecondary1.16e-065.86e-08LPG ProductionFormaldehydesecondary9.70e-074.92e-06US electric gridHydrofluoric acidmodel/secondary9.70e-074.92e-06US electric gridFormaldehydemodel/secondary9.70e-074.92e-06US electric gridFormaldehydemodel/secondary9.70e-074.92e-06US electric gridBenzenemodel/secondary9.70e-074.92e-06 <t< th=""><th>Process Group</th><th>Material</th><th>LCI Data Type</th><th>Terrestrial Toxicity (tox-kg)</th><th>% of Total</th></t<>	Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
US electric gridSeleniummodel/secondary9.28e-064.70e-07LPG ProductionVanadiumsecondary9.04e-064.58e-07LPG ProductionBenzenesecondary7.40e-063.75e-07LPG ProductionMethanesecondary7.24e-063.67e-07US electric gridVanadiummodel/secondary6.98e-063.54e-07LPG ProductionSulfur oxidessecondary6.98e-063.50e-07LPG ProductionSulfur oxidessecondary5.78e-062.93e-07LPG ProductionNitrogen oxidessecondary4.94e-062.93e-07LPG ProductionNitrogen oxidessecondary4.69e-062.37e-07CRT landfillingHydrochloric acidprimary4.69e-062.37e-07LPG ProductionArsenicsecondary1.35e-066.82e-08US electric gridHydrofluoric acidmodel/secondary1.35e-066.82e-08LPG ProductionFormaldehydesecondary1.11e-065.64e-08LPG ProductionFormaldehydesecondary9.72e-074.92e-06US electric gridHydrofluoric acidmodel/secondary9.70e-074.91e-06US electric gridFormaldehydemodel/secondary9.70e-074.91e-06US electric gridFormaldehydemodel/secondary9.70e-074.91e-06US electric gridBenzenemodel/secondary9.33e-071.99e-06US electric gridBenzenemodel/secondary9.39e-073.53e-06 <td>CRT landfilling</td> <td>Barium cmpds</td> <td>primary</td> <td>1.01e-05</td> <td>5.13e-07</td>	CRT landfilling	Barium cmpds	primary	1.01e-05	5.13e-07
LPG ProductionVanatiumsecondary9.04e-064.58e-07LPG ProductionBenzenesecondary7.40e-063.75e-07LPG ProductionMethanesecondary7.24e-063.67e-07US electric gridVanatiummodel/secondary6.98e-063.54e-07LPG ProductionSulfur oxidessecondary6.92e-063.50e-07US electric gridPM-10model/secondary5.78e-062.93e-07LPG ProductionNitrogen oxidessecondary4.94e-062.51e-07CRT landfillingHydrochloric acidprimary4.69e-062.37e-07CRT landfillingAmmoniaprimary2.41e-061.22e-07LPG ProductionArsenicsecondary1.78e-069.03e-08US electric gridHydrofluoric acidmodel/secondary1.35e-066.82e-08LPG ProductionFormaldehydesecondary1.11e-065.64e-08US electric gridPMsecondary9.70e-074.91e-08US electric gridFormaldehydemodel/secondary9.70e-074.91e-08US electric gridBenzenemodel/secondary9.70e-074.91e-08US electric gridBenzenemodel/secondary9.73e-071.99e-08US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-08US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08	US electric grid	Selenium	model/secondary	9.28e-06	4.70e-07
LPG ProductionBenzenesecondary7.40e-063.75e-07LPG ProductionMethanesecondary7.24e-063.67e-07US electric gridVanadiummodel/secondary6.98e-063.54e-07LPG ProductionSulfur oxidessecondary6.92e-063.50e-07US electric gridPM-10model/secondary5.78e-062.93e-07LPG ProductionNitrogen oxidessecondary4.94e-062.51e-07CRT landfillingHydrochloric acidprimary4.69e-062.37e-07CRT landfillingAmmoniaprimary2.41e-061.22e-07LPG ProductionArsenicsecondary1.35e-066.82e-08US electric gridHydrofluoric acidmodel/secondary1.35e-066.82e-08LPG ProductionFormaldehydesecondary1.11e-065.64e-08US electric gridFormaldehydesecondary9.72e-074.92e-08US electric gridFormaldehydemodel/secondary9.70e-074.91e-08CRT landfillingCadmium empdsprimary9.72e-074.92e-08US electric gridFormaldehydemodel/secondary9.70e-074.91e-08CRT landfillingCadmium empdssecondary9.70e-074.91e-08US electric gridBenzenemodel/secondary9.39e-073.53e-06US electric gridBenzenemodel/secondary3.93e-073.53e-06US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08 <td>LPG Production</td> <td>Vanadium</td> <td>secondary</td> <td>9.04e-06</td> <td>4.58e-07</td>	LPG Production	Vanadium	secondary	9.04e-06	4.58e-07
LPG ProductionMethanesecondary7.24e-063.67e-07US electric gridVanadiummodel/secondary6.98e-063.54e-07LPG ProductionSulfur oxidessecondary6.92e-063.50e-07US electric gridPM-10model/secondary5.78e-062.93e-07LPG ProductionNitrogen oxidessecondary4.94e-062.51e-07CRT landfillingHydrochloric acidprimary4.69e-062.37e-07LPG ProductionArsenicsecondary1.78e-069.03e-06LPG ProductionArsenicsecondary1.78e-069.03e-06US electric gridHydrofluoric acidmodel/secondary1.35e-066.82e-08LPG ProductionFormaldehydesecondary1.16e-065.86e-08LPG ProductionPMsecondary1.11e-065.64e-08US electric gridFormaldehydemodel/secondary9.72e-074.92e-08US electric gridFormaldehydemodel/secondary9.72e-074.92e-08US electric gridBenzenemodel/secondary9.73e-073.53e-08US electric gridBenzenemodel/secondary6.98e-073.53e-08US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-08US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08	LPG Production	Benzene	secondary	7.40e-06	3.75e-07
US electric gridVanadiummodel/secondary6.98e-063.54e-07LPG ProductionSulfur oxidessecondary6.92e-063.50e-07US electric gridPM-10model/secondary5.78e-062.93e-07LPG ProductionNitrogen oxidessecondary4.94e-062.51e-07CRT landfillingHydrochloric acidprimary4.69e-062.37e-07CRT landfillingAmmoniaprimary2.41e-061.22e-07LPG ProductionArsenicsecondary1.78e-069.03e-06US electric gridHydrofluoric acidmodel/secondary1.35e-066.82e-08LPG ProductionFormaldehydesecondary1.16e-065.86e-08LPG ProductionFormaldehydesecondary1.11e-065.64e-08LPG ProductionPMsecondary9.72e-074.92e-08US electric gridFormaldehydemodel/secondary9.72e-074.92e-08US electric gridFormaldehydemodel/secondary9.70e-074.91e-08US electric gridBenzenemodel/secondary9.73e-074.91e-08US electric gridBenzenemodel/secondary9.38e-073.53e-08US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-08US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08	LPG Production	Methane	secondary	7.24e-06	3.67e-07
LPG ProductionSulfur oxidessecondary6.92e-063.50e-07US electric gridPM-10model/secondary5.78e-062.93e-07LPG ProductionNitrogen oxidessecondary4.94e-062.51e-07CRT landfillingHydrochloric acidprimary4.69e-062.37e-07CRT landfillingAmmoniaprimary2.41e-061.22e-07LPG ProductionArsenicsecondary1.78e-069.03e-08US electric gridHydrofluoric acidmodel/secondary1.35e-066.82e-08LPG ProductionFormaldehydesecondary1.16e-065.86e-08LPG ProductionPMsecondary1.11e-065.64e-08LPG ProductionPMsecondary9.72e-074.92e-08US electric gridFormaldehydemodel/secondary9.70e-074.91e-08CRT landfillingCadmium empdsprimary9.72e-074.92e-08US electric gridFormaldehydemodel/secondary9.70e-074.91e-08US electric gridBenzenemodel/secondary9.38e-073.53e-08US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-08US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08	US electric grid	Vanadium	model/secondary	6.98e-06	3.54e-07
US electric gridPM-10model/secondary5.78e-062.93e-07LPG ProductionNitrogen oxidessecondary4.94e-062.51e-07CRT landfillingHydrochloric acidprimary4.69e-062.37e-07CRT landfillingAmmoniaprimary2.41e-061.22e-07LPG ProductionArsenicsecondary1.78e-069.03e-08US electric gridHydrofluoric acidmodel/secondary1.35e-066.82e-08LPG ProductionFormaldehydesecondary1.11e-065.64e-08LPG ProductionPMsecondary1.11e-065.64e-08LPG ProductionPMsecondary9.72e-074.92e-08US electric gridFormaldehydemodel/secondary9.70e-074.91e-08CRT landfillingCadmium empdssecondary9.70e-074.91e-08US electric gridBenzenemodel/secondary9.41e-074.77e-08US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-08US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08	LPG Production	Sulfur oxides	secondary	6.92e-06	3.50e-07
LPG ProductionNitrogen oxidessecondary4.94e-062.51e-0'CRT landfillingHydrochloric acidprimary4.69e-062.37e-0'CRT landfillingAmmoniaprimary2.41e-061.22e-0'LPG ProductionArsenicsecondary1.78e-069.03e-08US electric gridHydrofluoric acidmodel/secondary1.35e-066.82e-08LPG ProductionFormaldehydesecondary1.16e-065.86e-08LPG ProductionPMsecondary1.11e-065.64e-08CRT landfillingCadmium empdsprimary9.72e-074.92e-08US electric gridFormaldehydemodel/secondary9.70e-074.91e-08CRT IncinerationCadmium empdssecondary9.41e-074.77e-08US electric gridBenzenemodel/secondary6.98e-073.53e-08US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-08US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08	US electric grid	PM-10	model/secondary	5.78e-06	2.93e-07
CRT landfillingHydrochloric acidprimary4.69e-062.37e-07CRT landfillingAmmoniaprimary2.41e-061.22e-07LPG ProductionArsenicsecondary1.78e-069.03e-08US electric gridHydrofluoric acidmodel/secondary1.35e-066.82e-08LPG ProductionFormaldehydesecondary1.16e-065.86e-08LPG ProductionPMsecondary1.11e-065.64e-08CRT landfillingCadmium empdsprimary9.72e-074.92e-08US electric gridFormaldehydemodel/secondary9.70e-074.91e-08CRT landfillingCadmium empdssecondary9.70e-074.91e-08US electric gridBenzenemodel/secondary9.41e-074.77e-08US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-08US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08	LPG Production	Nitrogen oxides	secondary	4.94e-06	2.51e-07
CRT landfillingAmmoniaprimary2.41e-061.22e-07LPG ProductionArsenicsecondary1.78e-069.03e-08US electric gridHydrofluoric acidmodel/secondary1.35e-066.82e-08LPG ProductionFormaldehydesecondary1.16e-065.86e-08LPG ProductionPMsecondary1.11e-065.64e-08CRT landfillingCadmium empdsprimary9.72e-074.92e-08US electric gridFormaldehydemodel/secondary9.70e-074.91e-08CRT IncinerationCadmium empdssecondary9.41e-074.77e-08US electric gridBenzenemodel/secondary6.98e-073.53e-08US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-08US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08	CRT landfilling	Hydrochloric acid	primary	4.69e-06	2.37e-07
LPG ProductionArsenicsecondary1.78e-069.03e-04US electric gridHydrofluoric acidmodel/secondary1.35e-066.82e-04LPG ProductionFormaldehydesecondary1.16e-065.86e-04LPG ProductionPMsecondary1.11e-065.64e-04CRT landfillingCadmium cmpdsprimary9.72e-074.92e-04US electric gridFormaldehydemodel/secondary9.70e-074.91e-06CRT IncinerationCadmium cmpdssecondary9.70e-074.91e-06US electric gridBenzenemodel/secondary9.41e-074.77e-08US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-08US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08	CRT landfilling	Ammonia	primary	2.41e-06	1.22e-07
US electric gridHydrofluoric acidmodel/secondary1.35e-066.82e-04LPG ProductionFormaldehydesecondary1.16e-065.86e-04LPG ProductionPMsecondary1.11e-065.64e-04CRT landfillingCadmium cmpdsprimary9.72e-074.92e-04US electric gridFormaldehydemodel/secondary9.70e-074.91e-06CRT IncinerationCadmium cmpdssecondary9.41e-074.77e-08US electric gridBenzenemodel/secondary6.98e-073.53e-08US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-08US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08	LPG Production	Arsenic	secondary	1.78e-06	9.03e-08
LPG ProductionFormaldehydesecondary1.16e-065.86e-04LPG ProductionPMsecondary1.11e-065.64e-08CRT landfillingCadmium empdsprimary9.72e-074.92e-08US electric gridFormaldehydemodel/secondary9.70e-074.91e-08CRT IncinerationCadmium empdssecondary9.41e-074.77e-08US electric gridBenzenemodel/secondary6.98e-073.53e-08US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-08US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08	US electric grid	Hydrofluoric acid	model/secondary	1.35e-06	6.82e-08
LPG ProductionPMsecondary1.11e-065.64e-04CRT landfillingCadmium cmpdsprimary9.72e-074.92e-08US electric gridFormaldehydemodel/secondary9.70e-074.91e-08CRT IncinerationCadmium cmpdssecondary9.41e-074.77e-08US electric gridBenzenemodel/secondary6.98e-073.53e-08US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-08US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08	LPG Production	Formaldehyde	secondary	1.16e-06	5.86e-08
CRT landfillingCadmium empdsprimary9.72e-074.92e-04US electric gridFormaldehydemodel/secondary9.70e-074.91e-03CRT IncinerationCadmium empdssecondary9.41e-074.77e-03US electric gridBenzenemodel/secondary6.98e-073.53e-03US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-03US electric gridNitrous oxidemodel/secondary3.40e-071.72e-03	LPG Production	PM	secondary	1.11e-06	5.64e-08
US electric gridFormaldehydemodel/secondary9.70e-074.91e-00CRT IncinerationCadmium cmpdssecondary9.41e-074.77e-08US electric gridBenzenemodel/secondary6.98e-073.53e-08US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-08US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08	CRT landfilling	Cadmium cmpds	primary	9.72e-07	4.92e-08
CRT IncinerationCadmium cmpdssecondary9.41e-074.77e-08US electric gridBenzenemodel/secondary6.98e-073.53e-08US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-08US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08	US electric grid	Formaldehyde	model/secondary	9.70e-07	4.91e-08
US electric gridBenzenemodel/secondary6.98e-073.53e-08US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-08US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08	CRT Incineration	Cadmium cmpds	secondary	9.41e-07	4.77e-08
US electric gridPhosphorus (yellow or white)model/secondary3.93e-071.99e-08US electric gridNitrous oxidemodel/secondary3.40e-071.72e-08	US electric grid	Benzene	model/secondary	6.98e-07	3.53e-08
US electric grid Nitrous oxide model/secondary 3.40e-07 1.72e-08	US electric grid	Phosphorus (yellow or white)	model/secondary	3.93e-07	1.99e-08
	US electric grid	Nitrous oxide	model/secondary	3.40e-07	1.72e-08
LPG Production Hydrochloric acid secondary 3.18e-07 1.61e-08	LPG Production	Hydrochloric acid	secondary	3.18e-07	1.61e-08
CRT landfilling Hydrogen sulfide primary 2.30e-07 1.17e-08	CRT landfilling	Hydrogen sulfide	primary	2.30e-07	1.17e-08
US electric grid Zinc (elemental) model/secondary 1.81e-07 9.19e-09	US electric grid	Zinc (elemental)	model/secondary	1.81e-07	9.19e-09
US electric grid Antimony model/secondary 1.48e-07 7.52e-09	US electric grid	Antimony	model/secondary	1.48e-07	7.52e-09
LPG Production Nitrous oxide secondary 1.40e-07 7.09e-09	LPG Production	Nitrous oxide	secondary	1.40e-07	7.09e-09
LPG Production Phosphorus (yellow or white) secondary 8.47e-08 4.29e-09	LPG Production	Phosphorus (yellow or white)	secondary	8.47e-08	4.29e-09
LPG Production Fluorides (F-) secondary 6.48e-08 3.28e-09	LPG Production	Fluorides (F-)	secondary	6.48e-08	3.28e-09
LPG Production Selenium secondary 6.40e-08 3.24e-09	LPG Production	Selenium	secondary	6.40e-08	3.24e-09
US electric grid Molybdenum model/secondary 4.97e-08 2.52e-09	US electric grid	Molybdenum	model/secondary	4.97e-08	2.52e-09
US electric grid 2-Chloroacetophenone model/secondary 4.32e-08 2.19e-09	US electric grid	2-Chloroacetophenone	model/secondary	4.32e-08	2.19e-09
US electric grid Bromomethane model/secondary 4.27e-08 2.16e-09	US electric grid	Bromomethane	model/secondary	4.27e-08	2.16e-09
LPG Production Ammonia secondary 3.23e-08 1.63e-09	LPG Production	Ammonia	secondary	3.23e-08	1.63e-09
CRT landfilling Selenium primary 3.21e-08 1.62e-09	CRT landfilling	Selenium	primary	3.21e-08	1.62e-09
CRT landfilling Chromium (VI) primary 2.84e-08 1.44e-09	CRT landfilling	Chromium (VI)	primary	2.84e-08	1.44e-09
LPG Production Hydrogen sulfide secondary 2.65e-08 1.34e-09	LPG Production	Hydrogen sulfide	secondary	2.65e-08	1.34e-09
US electric grid Cyanide (-1) model/secondary 2.24e-08 1.14e-09	US electric grid	Cyanide (-1)	model/secondary	2.24e-08	1.14e-09
CRT Incineration Carbon tetrachloride secondary 1.80e-08 9.11e-10	CRT Incineration	Carbon tetrachloride	secondary	1.80e-08	9.11e-10
US electric grid 2,3,7,8-TCDD model/secondary 1.70e-08 8.59e-10	US electric grid	2,3,7,8-TCDD	model/secondary	1.70e-08	8.59e-10
CRT landfilling Carbon tetrachloride primary 1.68e-08 8.50e-10	CRT landfilling	Carbon tetrachloride	primary	1.68e-08	8.50e-10
US electric grid Nickel model/secondary 1.63e-08 8.26e-10	US electric grid	Nickel	model/secondary	1.63e-08	8.26e-10
LPG Production Molybdenum secondary 1.42e-08 7.20e-10	LPG Production	Molybdenum	secondary	1.42e-08	7.20e-10
US electric grid Naphthalene model/secondary 1.35e-08 6.85e-10	US electric grid	Naphthalene	model/secondary	1.35e-08	6.85e-10
CRT landfilling Chloroform primary 1.30e-08 6.59e-10	CRT landfilling	Chloroform	primary	1.30e-08	6.59e-10
US electric grid Barium model/secondary 1.17e-08 5.90e-10	US electric grid	Barium	model/secondary	1.17e-08	5.90e-10
CRT landfilling Mercury compounds primary 1.15e-08 5.84e-10	CRT landfilling	Mercury compounds	primary	1.15e-08	5.84e-10
CRT Incineration Mercury compounds secondary 1.13e-08 5.72e-10	CRT Incineration	Mercury compounds	secondary	1.13e-08	5.72e-10
LPG Production Zinc (elemental) secondary 1.04e-08 5.24e-10	LPG Production	Zinc (elemental)	secondary	1.04e-08	5.24e-10
LPG Production Hydrofluoric acid secondary 8.66e-09 4.39e-10	LPG Production	Hydrofluoric acid	secondary	8.66e-09	4.39e-10
CRT landfilling Toluene primary 8.22e-09 4.17e-10	CRT landfilling	Toluene	primary	8.22e-09	4.17e-10

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
US electric grid	Carbon disulfide	model/secondary	8.01e-09	4.06e-10
LPG Production	Ethane	secondary	6.70e-09	3.39e-10
US electric grid	Benzyl chloride	model/secondary	6.28e-09	3.18e-10
LPG Production	Phenol	secondary	6.17e-09	3.13e-10
LPG Production	Pentane	secondary	5.62e-09	2.84e-10
US electric grid	Chloroform	model/secondary	4.88e-09	2.47e-10
US electric grid	Manganese	model/secondary	4.56e-09	2.31e-10
LPG Production	Hexane	secondary	3.89e-09	1.97e-10
US electric grid	Chromium (VI)	model/secondary	3.44e-09	1.74e-10
US electric grid	Acrolein	model/secondary	2.60e-09	1.32e-10
LPG Production	Nickel	secondary	2.51e-09	1.27e-10
US electric grid	Copper	model/secondary	2.24e-09	1.13e-10
US electric grid	Methyl chloride	model/secondary	2.11e-09	1.07e-10
US electric grid	Fluoride	model/secondary	1.95e-09	9.90e-11
LPG Production	PM-10	secondary	1.94e-09	9.84e-11
LPG Production	Antimony	secondary	1.85e-09	9.35e-11
CRT landfilling	Xylene (mixed isomers)	primary	1.69e-09	8.59e-11
US electric grid	Methyl hydrazine	model/secondary	1.53e-09	7.73e-11
US electric grid	Lead	model/secondary	1.28e-09	6.46e-11
LPG Production	Aluminum (+3)	secondary	1.22e-09	6.19e-11
US electric grid	Cobalt	model/secondary	1.22e-09	6.16e-11
CRT landfilling	Tetrachloroethylene	primary	1.20e-09	6.07e-11
US electric grid	Propionaldehyde	model/secondary	1.17e-09	5.93e-11
US electric grid	Acetaldehyde	model/secondary	1.17e-09	5.93e-11
CRT landfilling	1,2-Dichloroethane	primary	9.32e-10	4.72e-11
US electric grid	Di(2-ethylhexyl)phthalate	model/secondary	9.00e-10	4.56e-11
LPG Production	Chromium (VI)	secondary	8.70e-10	4.41e-11
CRT Incineration	Lead cmpds	secondary	8.10e-10	4.11e-11
CRT landfilling	Lead cmpds	primary	7.89e-10	4.00e-11
US electric grid	Mercury	model/secondary	7.52e-10	3.81e-11
CRT Incineration	Trichloroethylene	secondary	7.50e-10	3.80e-11
CRT landfilling	Trichloroethylene	primary	6.99e-10	3.54e-11
US electric grid	Hexane	model/secondary	6.01e-10	3.05e-11
LPG Production	Nitrate	secondary	5.23e-10	2.65e-11
US electric grid	Cadmium	model/secondary	4.82e-10	2.44e-11
CRT landfilling	Ethylbenzene	primary	4.44e-10	2.25e-11
US electric grid	Dimethyl sulfate	model/secondary	4.31e-10	2.18e-11
US electric grid	Toluene	model/secondary	4.25e-10	2.15e-11
US electric grid	Isophorone	model/secondary	4.13e-10	2.09e-11
US electric grid	Methyl ethyl ketone	model/secondary	3.33e-10	1.69e-11
US electric grid	Tetrachloroethylene	model/secondary	3.28e-10	1.66e-11
LPG Production	Copper	secondary	3.01e-10	1.52e-11
US electric grid	Methyl methacrylate	model/secondary	2.85e-10	1.44e-11
LPG Production	2-Chloroacetophenone	secondary	2.78e-10	1.41e-11
LPG Production	Dimethylbenzanthracene	secondary	2.75e-10	1.40e-11
LPG Production	Bromomethane	secondary	2.75e-10	1.39e-11
US electric grid	1.2-Dichloroethane	model/secondary	2.37e-10	1.20e-11
US electric grid	Bromoform	model/secondary	2.33e-10	1.18e-11
US electric grid	Dichloromethane	model/secondary	2.25e-10	1.14e-11
LPG Production	Naphthalene	secondary	2.23e-10	1.13e-11
CRT landfilling	Dichloromethane	primary	1.94e-10	9.85e-12
		1 ··· 2		

LPG ProductionManganesesecondary1.93e-109.76US electric gridBerylliummodel/secondary1.91e-109.67US electric gridChlorobenzenemodel/secondary1.88e-109.52LPG ProductionBariumsecondary1.61e-108.15US electric grid2,4-Dinitrotoluenemodel/secondary1.49e-107.57	6e-12 7e-12 2e-12 5e-12 7e-12 0e-12 1e-12 4e-12 0e-12 5e-12 5e-12
US electric gridBerylliummodel/secondary1.91e-109.67US electric gridChlorobenzenemodel/secondary1.88e-109.52LPG ProductionBariumsecondary1.61e-108.15US electric grid2,4-Dinitrotoluenemodel/secondary1.49e-107.57	7e-12 2e-12 5e-12 7e-12 0e-12 1e-12 4e-12 0e-12 5e-12 4e-12
US electric gridChlorobenzenemodel/secondary1.88e-109.52LPG ProductionBariumsecondary1.61e-108.15US electric grid2,4-Dinitrotoluenemodel/secondary1.49e-107.57	2e-12 5e-12 7e-12 0e-12 1e-12 4e-12 0e-12 5e-12 4e-12
LPG ProductionBariumsecondary1.61e-108.15US electric grid2,4-Dinitrotoluenemodel/secondary1.49e-107.57	5e-12 7e-12 0e-12 1e-12 4e-12 0e-12 5e-12 4e-12
US electric grid 2,4-Dinitrotoluene model/secondary 1.49e-10 7.57	7e-12 De-12 1e-12 4e-12 De-12 5e-12 4e-12
	0e-12 1e-12 4e-12 0e-12 5e-12 4e-12
LPG Production Silicon secondary 1.46e-10 7.40	1e-12 4e-12 De-12 5e-12 4e-12
LPG Production Cyanide (-1) secondary 1.44e-10 7.31	4e-12 0e-12 5e-12 4e-12
LPG Production Barium cmpds secondary 1.37e-10 6.94	0e-12 5e-12 4e-12
LPG Production Lead secondary 1.07e-10 5.40	5e-12 4e-12
US electric grid Ethylbenzene model/secondary 7.41e-11 3.75	4e-12
CRT Incineration o-xylene secondary 5.21e-11 2.64	
LPG Production Carbon disulfide secondary 5.16e-11 2.61	1e-12
CRT landfilling Chromium (III) primary 4.83e-11 2.45	5e-12
LPG Production Benzyl chloride secondary 4.04e-11 2.05	5e-12
US electric grid Methyl tert-butyl ether model/secondary 3.74e-11 1.89	9e-12
LPG Production Aluminum (elemental) secondary 3.34e-11 1.69	9e-12
LPG Production Chloroform secondary 3.14e-11 1.59	9e-12
US electric grid Phenol model/secondary 2.85e-11 1.44	4e-12
US electric grid Styrene model/secondary 2.73e-11 1.38	8e-12
US electric grid Vinyl acetate model/secondary 2.66e-11 1.35	5e-12
US electric grid Phenanthrene model/secondary 2.51e-11 1.27	7e-12
US electric grid Xylene (mixed isomers) model/secondary 2.21e-11 1.12	2e-12
LPG Production Cobalt secondary 2.17e-11 1.10	0e-12
US electric grid Chromium (III) model/secondary 1.97e-11 9.99	9e-13
LPG Production Acrolein secondary 1.67e-11 8.48	8e-13
LPG Production Methyl chloride secondary 1.36e-11 6.87	7e-13
LPG Production Beryllium secondary 1.21e-11 6.14	4e-13
US electric grid 1,1,1-Trichloroethane model/secondary 1.08e-11 5.50	0e-13
US electric grid Ethylene dibromide model/secondary 1.08e-11 5.45	5e-13
LPG Production Methyl hydrazine secondary 9.82e-12 4.97	7e-13
LPG Production Acetaldehyde secondary 7.54e-12 3.82	2e-13
LPG Production Propionaldehyde secondary 7.54e-12 3.82	2e-13
LPG Production Mercury secondary 7.50e-12 3.80	0e-13
US electric grid Ethyl Chloride model/secondary 7.19e-12 3.64	4e-13
LPG Production Cadmium secondary 6.99e-12 3.54	4e-13
US electric grid Cumene model/secondary 6.08e-12 3.08	8e-13
LPG Production Cadmium cmpds secondary 5.99e-12 3.04	4e-13
LPG Production Di(2-ethylhexyl)phthalate secondary 5.79e-12 2.93	3e-13
LPG Production Toluene secondary 5.38e-12 2.72	2e-13
US electric grid Acetophenone model/secondary 3.79e-12 1.92	2e-13
US electric grid Biphenyl model/secondary 3.63e-12 1.8 ⁴	4e-13
LPG Production 1,4-Dichlorobenzene secondary 3.08e-12 1.56	5e-13
CRT Incineration Vinvl chloride secondary 2.97e-12 1.51	1e-13
CRT landfilling Vinvl chloride primary 2.77e-12 1.41	1e-13
LPG Production Dimethyl sulfate secondary 2.77e-12 1.40	0e-13
LPG Production Isophorone secondary 2.66e-12 1.35	5e-13
US electric grid Acenaphthylene model/secondary 2.26e-12 1.14	4e-13
LPG Production Methyl ketone secondary 2.14e-12 1.09	9e-13
LPG Production Tetrachloroethylene secondary 2.11e-12 1.07	7e-13
US electric grid Acenaphthene model/secondary 1.93e-12 9.79	9e-14

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
LPG Production	Methyl methacrylate	secondary	1.83e-12	9.28e-14
LPG Production	1,2-Dichloroethane	secondary	1.53e-12	7.74e-14
LPG Production	Bromoform	secondary	1.50e-12	7.59e-14
LPG Production	Chromium (III)	secondary	1.48e-12	7.51e-14
LPG Production	Dichloromethane	secondary	1.45e-12	7.32e-14
LPG Production	Chlorobenzene	secondary	1.21e-12	6.13e-14
LPG Production	Aromatic hydrocarbons	secondary	1.21e-12	6.12e-14
LPG Production	Copper (+1 & +2)	secondary	1.13e-12	5.73e-14
US electric grid	Benzo[b,j,k]fluoranthene	model/secondary	1.06e-12	5.39e-14
US electric grid	Chrysene	model/secondary	1.02e-12	5.18e-14
LPG Production	2,4-Dinitrotoluene	secondary	9.62e-13	4.87e-14
US electric grid	Benzo[a]anthracene	model/secondary	9.28e-13	4.70e-14
US electric grid	Fluorene	model/secondary	8.00e-13	4.05e-14
US electric grid	Indeno(1,2,3-cd)pyrene	model/secondary	6.59e-13	3.34e-14
US electric grid	Fluoranthene	model/secondary	6.39e-13	3.24e-14
LPG Production	o-xylene	secondary	5.92e-13	3.00e-14
US electric grid	Pyrene	model/secondary	5.29e-13	2.68e-14
LPG Production	Ethylbenzene	secondary	4.80e-13	2.43e-14
US electric grid	o-xylene	model/secondary	3.80e-13	1.92e-14
US electric grid	Benzo[g,h,i]perylene	model/secondary	3.61e-13	1.83e-14
US electric grid	Benzo[a]pyrene	model/secondary	3.41e-13	1.73e-14
US electric grid	2-Methylnaphthalene	model/secondary	2.67e-13	1.35e-14
LPG Production	Methyl tert-butyl ether	secondary	2.41e-13	1.22e-14
US electric grid	5-Methyl chrysene	model/secondary	1.97e-13	1.00e-14
LPG Production	Phenanthrene	secondary	1.96e-13	9.91e-15
LPG Production	Styrene	secondary	1.76e-13	8.89e-15
LPG Production	Vinyl acetate	secondary	1.71e-13	8.68e-15
LPG Production	3-Methylcholanthrene	secondary	1.62e-13	8.19e-15
LPG Production	Zinc (+2)	secondary	1.01e-13	5.13e-15
US electric grid	Dibenzo[a,h]anthracene	model/secondary	8.75e-14	4.43e-15
LPG Production	Ethylene dibromide	secondary	6.93e-14	3.51e-15
LPG Production	1,1,1-Trichloroethane	secondary	6.53e-14	3.31e-15
LPG Production	2-Methylnaphthalene	secondary	5.18e-14	2.63e-15
LPG Production	Ethyl Chloride	secondary	4.63e-14	2.35e-15
LPG Production	Chlorine	secondary	4.59e-14	2.33e-15
LPG Production	Cumene	secondary	3.92e-14	1.98e-15
LPG Production	Acetophenone	secondary	2.44e-14	1.23e-15
LPG Production	Biphenyl	secondary	2.34e-14	1.18e-15
US electric grid	Anthracene	model/secondary	2.32e-14	1.17e-15
LPG Production	Acenaphthylene	secondary	1.85e-14	9.36e-16
LPG Production	Acenaphthene	secondary	1.54e-14	7.80e-16
LPG Production	Nickel cmpds	secondary	1.20e-14	6.07e-16
LPG Production	Benzo[a]anthracene	secondary	1.08e-14	5.46e-16
LPG Production	Chrysene	secondary	1.03e-14	5.24e-16
LPG Production	Lead cmpds	secondary	1.01e-14	5.10e-16
LPG Production	Indeno(1,2,3-cd)pyrene	secondary	8.02e-15	4.06e-16
LPG Production	Benzo[b,j,k]fluoranthene	secondary	6.35e-15	3.22e-16
LPG Production	Mercury compounds	secondary	6.10e-15	3.09e-16
LPG Production	Benzo[a]pyrene	secondary	5.90e-15	2.99e-16
LPG Production	Fluorene	secondary	5.70e-15	2.89e-16

LPG Production Pyrone secondary 4.78-15 2.780-16 LPG Production Baroz (a)Lipelyne secondary 4.78-15 2.743-16 LPG Production HALON-1301 secondary 4.85-15 2.78-16 LPG Production Dibarzo (a) jubaranthene secondary 4.31-15 2.18-16 LPG Production Dibarzo (a) jubaranthene secondary 1.27-15 6.44-17 LPG Production Halogenated mater (reganic) secondary 2.32-17 1.278-15 6.44-17 LPG Production Halogenated Inforcentors (unspecified) secondary 2.32-17 1.278-19 Natural Gas Prod. Halogenated Inforcentors (unspecified) secondary 2.35-18 1.358-19 Natural Gas Prod. Halogenated Inforcentors (unspecified) secondary -3.18-16 -5.08-17 Natural Gas Prod. Halogenated Inforcentors (unspecified) secondary -1.18-14 -5.08-17 Natural Gas Prod. HALON-1301 secondary -1.18-14 -5.08-17 Natural Gas Prod. HALON-1301 secondary -1.18-14<	Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
LPG Production Buzzoj Lul perylene secondary 4.78-15 2.248-16 LPG Production Fluczendhene secondary 4.38-15 2.228-16 LPG Production Buzzoj Uj fluzzambene secondary 3.06-15 2.18-16 LPG Production Dihezzoj Laj landmanene secondary 3.06-15 5.158-16 LPG Production Talkop secondary 1.27e-15 6.44e-17 LPG Production Antimene secondary 1.27e-15 5.04e-17 LPG Production Antimene secondary 2.14e-16 1.09e-17 LPG Production Antimenen secondary 2.14e-16 1.09e-17 LPG Production Halogenated hydrocarbons (unspecified) secondary -2.39e-18 1.38e-19 Naturi Gas Prod. Halogenated hydrocarbons (unspecified) secondary -1.18e-15 -2.30e-18 Fuel Oil 14 Prod. Halogenated hydrocarbons (unspecified) secondary -1.18e-15 -5.99e-16 Naturi Gas Prod. Halogenated hydrocarbons (unspecified) secondary -1.18e-14 -5.19e-16 <	LPG Production	Pyrene	secondary	4.93e-15	2.50e-16
LPG Ponduction HACM-1801 secondary 4458-15 2.238-16 LPG Ponduction Benzo[b]hasramhene secondary 4.318-15 2.188-16 LPG Ponduction Dibenzo[a,h]anthracne secondary 1.378-16 5.484-15 LPG Ponduction Secondary 1.378-16 5.484-15 5.318-17 LPG Ponduction Anthracne secondary 4.386-16 2.328-16 LPG Ponduction Halogenated mater (organic) secondary 4.586-16 2.328-17 LPG Ponduction Halogenated hydrocarbons (unspecified) secondary 2.528-18 1.282-19 Natural Gas Pod. Halogenated hydrocarbons (unspecified) secondary 4.358-17 2.220-18 Palo OH #4 Pod. Halogenated hydrocarbons (unspecified) secondary 4.158-15 2.261-16 Natural Gas Pod. Halogenated hydrocarbons (unspecified) secondary 4.158-15 2.261-16 Natural Gas Pod. Halogenated mater (organic) secondary 4.186-14 -4.326-15 Natural Gas Pod. Halogenated mater (organic) secondary 4.186-14 <td>LPG Production</td> <td>Benzo[g,h,i]perylene</td> <td>secondary</td> <td>4.79e-15</td> <td>2.43e-16</td>	LPG Production	Benzo[g,h,i]perylene	secondary	4.79e-15	2.43e-16
LPG Production HALON-1501 secondary 4.38x-15 2.22x-16 LPG Production Diberazi(a,h]anthracene secondary 3.06x-15 1.55x-16 LPG Production 5.Methyl chrysene secondary 1.01x-15 5.04x-17 LPG Production Halogenated matter (organic) secondary 1.01x-15 5.04x-17 US electric grid 2.3,7,8-TCDF model/secondary 2.14x-16 1.03x-17 US Production Anthracene secondary 2.14x-16 1.09x-17 US Production Halogenated hydrocarbons (unspecified) secondary -2.35x-18 1.15x-19 Natural Gas Prod. Halogenated hydrocarbons (unspecified) secondary -3.15x-16 -3.05x-17 Natural Gas Prod. Halogenated hydrocarbons (unspecified) secondary -1.15x-16 -3.05x-15 Natural Gas Prod. Halogenated hydrocarbons (unspecified) secondary -1.15x-16 -3.05x-15 Natural Gas Prod. Halogenated hydrocarbons (unspecified) secondary -1.15x-16 -3.05x-15 Natural Gas Prod. Halogenated hydrocarbons (unspecified)	LPG Production	Fluoranthene	secondary	4.65e-15	2.36e-16
LPC PoductionDimon Difthormlenesecondary4.18-152.18e-16LPG ProductionDiheray cal plantracenesecondary1.27e-156.44e-17LPG ProductionLalogenated natter (organic)secondary1.58e-162.35.01e-17US sletric grid2.37.57 CDFmodel/secondary4.58e-162.35.01e-17LPG ProductionHalogenated hydrocarbons (unspecified)secondary2.48e-162.350e-17LPG ProductionHalogenated hydrocarbons (unspecified)secondary2.48e-163.350e-17LPG ProductionHalogenated hydrocarbons (unspecified)secondary-1.35e-17-2.20e-18Patel OI 14 Prod.Halogenated hydrocarbons (unspecified)secondary-1.35e-16-3.62e-17Natural Gas Prod.Halogenated hydrocarbons (unspecified)secondary-1.18e-15-5.99e-17Natural Gas Prod.HALON-1301secondary-1.18e-14-7.195e-15Natural Gas Prod.Halogenated natter (organic)secondary-1.18e-14-7.195e-16Natural Gas Prod.Marcury compoundssecondary-1.18e-14-7.18e-16Natural Gas Prod.HALON-1301secondary-1.18e-14-7.18e-16Natural Gas Prod.Anthracenesecondary-1.48e-14-7.38e-16Natural Gas Prod.Anthracenesecondary-1.48e-14-7.38e-16Natural Gas Prod.HALON-1301secondary-1.28e-15-7.48e-15CRT IncinerationHALON-1301secondary-1.28e-15-7.48e-16Pado Di	LPG Production	HALON-1301	secondary	4.38e-15	2.22e-16
LPG Production Diberacja/halmacene secondary 3.06e-15 1.55e-16 LPG Production 4.Medya letryscne secondary 1.01e-15 5.10e-17 LPG Production Anthracene secondary 4.58e-16 2.32e-17 LPG Production Anthracene secondary 2.52e-18 1.58e-16 LPG Production Halogenated hydrocarbons (unspecified) secondary -2.95e-18 1.58e-16 CRT Insincation Halogenated hydrocarbons (unspecified) secondary -7.15e-16 3.62e-17 Vatural Gas Prod. Halogenated hydrocarbons (unspecified) secondary -7.15e-15 -3.62e-16 Natural Gas Prod. Halogenated mater (organic) secondary -7.15e-15 -3.63e-16 Natural Gas Prod. Mercury compounds secondary -1.41e-14 -7.13e-15 Natural Gas Prod. Mercury compounds secondary -1.41e-14 -2.43e-15 CRT Insinearion Halogenated mater (organic) secondary -4.41e-14 -2.45e-15 Natural Gas Prod. Anthracene secondary -4.41e-14	LPG Production	Benzo[b]fluoranthene	secondary	4.31e-15	2.18e-16
LPG Production 5-Methyl chrysne secondary 1.27e-15 6.44e-17 LPG Production Halogenated matter (organic) secondary 4.38e-16 2.37e-17 LPG Production Andiracene secondary 2.14e-16 1.09e-17 LPG Production Halogenated hydrocarbons (unspecific) secondary -2.35e-18 1.28e-19 Natural Gas Prod. Halogenated hydrocarbons (unspecific) secondary -7.15e-16 3.62e-17 Natural Gas Prod. Halogenated hydrocarbons (unspecific) secondary -1.18e-15 -5.99e-17 Natural Gas Prod. Halogenated hydrocarbons (unspecific) secondary -1.18e-15 -5.99e-17 Natural Gas Prod. Halogenated matter (organic) secondary -1.18e-14 -5.99e-17 Natural Gas Prod. Mercury compounds secondary -1.18e-14 -5.99e-16 Natural Gas Prod. Natural Gas Prod. Natural Gas Prod. -3.09e-16 -3.09e-16 Natural Gas Prod. Anthracene secondary -1.48e-14 -2.49e-15 CRT Incineration Halogenated matter (organic) <t< td=""><td>LPG Production</td><td>Dibenzo[a,h]anthracene</td><td>secondary</td><td>3.06e-15</td><td>1.55e-16</td></t<>	LPG Production	Dibenzo[a,h]anthracene	secondary	3.06e-15	1.55e-16
LPG ProductionHalogenated matter (organic)secondary1.01e-155.10e-17US electric grid2,3,7,8 TCDFmodel/secondary4.58e-162.32a-17LPG ProductionAndiracenesecondary2.55a-181.28e-19LPG ProductionHalogenated hydrocarbons (unspecified)secondary2.55e-181.28e-19CRT IncinerationHalogenated hydrocarbons (unspecified)secondary4.35e-172.20e-18Fuel Oil #4 Prod.Halogenated hydrocarbons (unspecified)secondary-1.18e-155.59e-17Natural Gas Prod.Halogenated nydrocarbons (unspecified)secondary-1.18e-155.59e-17Natural Gas Prod.Macor corganic)secondary-7.15e-16-3.63e-16Natural Gas Prod.Macor corganic)secondary-1.18e-14-5.99e-16Natural Gas Prod.Lad empdssecondary-1.18e-14-5.99e-16Natural Gas Prod.Nickel enpdssecondary-1.18e-14-2.43e-15Natural Gas Prod.Nickel enpdssecondary-1.48e-14-2.43e-15Natural Gas Prod.Anthracenesecondary-1.28e-13-4.19e-14Natural Gas Prod.Anthracenesecondary-2.8e-14-4.26e-15Natural Gas Prod.Anthracenesecondary-2.8e-14-4.26e-15Natural Gas Prod.Halogenated matter (organic)secondary-2.8e-14-4.26e-15Natural Gas Prod.Shethyl chrysenesecondary-2.8e-13-1.11e-14Fuel Oil #4 Prod.Benzolphijkintinenen	LPG Production	5-Methyl chrysene	secondary	1.27e-15	6.44e-17
US electric grid 2,3,7,8-TCDF model/secondary 4,88+16 2,32e-17 IPG Production Anbracenc secondary 2,32e-18 1,28e-19 Natural Gas Prod. Halogenated hydrocarbons (unspecified) secondary -2,55e-18 -1,30e-17 CRT Incineration Halogenated hydrocarbons (unspecified) secondary -4,35e-17 -2,20e-18 Fuel Oil #4 Prod. Halogenated hydrocarbons (unspecified) secondary -1,18e-16 -3,62e-17 Natural Gas Prod. HALON-1301 secondary -7,15e-16 -3,62e-16 Natural Gas Prod. Lead empds secondary -1,18e-14 -5,39e-16 Natural Gas Prod. Lead empds secondary -1,41e-14 -7,13e-16 ORT Incineration Halogenated matter (organic) secondary -1,41e-14 -7,13e-16 Natural Gas Prod. Anthracene secondary -1,41e-14 -7,13e-16 CRT Incineration Halogenated matter (organic) secondary -4,79e-14 -2,28e-13 CRT Incineration HALON-1301 secondary -2,86e-13 <td< td=""><td>LPG Production</td><td>Halogenated matter (organic)</td><td>secondary</td><td>1.01e-15</td><td>5.10e-17</td></td<>	LPG Production	Halogenated matter (organic)	secondary	1.01e-15	5.10e-17
LPG ProductionAnthracenesecondary2.14e-161.09e-17LPG ProductionHalogenated hydrocarbons (unspecified)secondary-2.52e-181.28e-19Nataral Gas Prod.Halogenated hydrocarbons (unspecified)secondary-2.55e-16-2.20e-17Natural Gas Prod.Halogenated hydrocarbons (unspecified)secondary-7.15e-16-3.62e-17Natural Gas Prod.Halogenated matter (organic)secondary-7.15e-16-3.62e-17Natural Gas Prod.Mcurey compoundssecondary-7.16e-15-3.63e-16Natural Gas Prod.Lead empdssecondary-1.18e-14-5.99e-16Natural Gas Prod.Lead empdssecondary-1.18e-14-5.99e-16Natural Gas Prod.Lead empdssecondary-1.18e-14-4.90e-16Natural Gas Prod.Halogenated matter (organic)secondary-1.48e-14-7.13e-16CRT IncinerationHALON-1301secondary-1.48e-144.05e-15CRT IncinerationHALON-1301secondary-1.48e-14-4.59e-16Natural Gas Prod.Anthracenesecondary-1.48e-13-6.49e-15CRT IncinerationNickel empdssecondary-2.48e-13-4.11e-14Led Oil # Prod.Anthracenesecondary-2.88e-13-4.11e-14Led Oil # Prod.Anthracenesecondary-2.88e-13-4.58e-14CRT IncinerationNickel empdssecondary-2.88e-13-4.58e-14CRT IncinerationDibenzo[A.J]anthracenesecondary-2.28e-1	US electric grid	2,3,7,8-TCDF	model/secondary	4.58e-16	2.32e-17
LPG ProductionHalogenated hydrocarbons (unspecified)secondary2.52e-181.28e-19Natural Gas Prod.Halogenated hydrocarbons (unspecified)secondary-4.35e-17Natural Gas Prod.Halogenated hydrocarbons (unspecified)secondary-1.51e-16Natural Gas Prod.Halogenated hydrocarbons (unspecified)secondary-1.18e-15Natural Gas Prod.HALON-1301secondary-1.18e-15-2.61e-16Natural Gas Prod.Lead empdssecondary-1.18e-14-5.99e-17Natural Gas Prod.Lead empdssecondary-1.18e-14-5.99e-16Natural Gas Prod.Halogenated matter (organic)secondary-1.18e-14-5.99e-16Natural Gas Prod.Matter (organic)secondary-1.41e-14-7.13e-16CRT IncinerationHALON-1301secondary-1.41e-14-7.13e-16Natural Gas Prod.Anthrecnesecondary-2.8ee-13-6.49e-15Natural Gas Prod.Anthrecnesecondary-2.8e-13-6.49e-15Natural Gas Prod.S-Methyl chrysenesecondary-2.8ee-13-1.45e-14CRT IncinerationDibenzo(a,h)anthrecnesecondary-2.8ee-13-2.8e-14Puel Oil # Prod.Barbyl, chrysenesecondary-2.8ee-13-2.8e-14CRT IncinerationDibenzo(a,h)anthrecnesecondary-2.8ee-13-3.37e-14Vatural Gas Prod.Fluorenesecondary-3.2ee-14-2.28e-14CRT IncinerationDibenzo(a,h)anthrecnesecondary-2.8ee-13-3	LPG Production	Anthracene	secondary	2.14e-16	1.09e-17
Natural Gas Prod.Halogenated hydrocarbons (unspecified)secondary4.35e-17-2.20e-18CRT IncinerationHalogenated hydrocarbons (unspecified)secondary-1.15e-16-3.62e-17Natural Gas Prod.Halogenated hydrocarbons (unspecified)secondary-1.15e-15-5.99e-17Natural Gas Prod.HALON-1301secondary-7.16e-15-3.62e-16Natural Gas Prod.Mercury compoundssecondary-7.16e-15-3.63e-16Natural Gas Prod.Nickel emptssecondary-1.18e-14-5.99e-16Natural Gas Prod.Nickel emptssecondary-1.41e-14-7.13e-16CRT IncinerationHalogenated matter (organic)secondary-1.84e-14-9.30e-16Natural Gas Prod.Anthracenesecondary-1.84e-14-4.26e-15Natural Gas Prod.Anthracenesecondary-1.75e-13-6.46e-15CRT IncinerationHALON-1301secondary-2.88e-13-1.11e-14Fuel Oil #4 Prod.Anthracenesecondary-2.28e-13-1.46e-15Natural Gas Prod.S-Methyl chrysenesecondary-2.28e-13-1.47e-14Fuel Oil #4 Prod.Anthracenesecondary-2.28e-13-1.47e-14Fuel Oil #4 Prod.Halogenated matter (organic)secondary-2.28e-13-1.47e-14Fuel Oil #4 Prod.Benzo[h]nhtracenesecondary-2.28e-13-1.47e-14Fuel Oil #4 Prod.Benzo[h]nhtracenesecondary-2.28e-13-1.47e-14Fuel Oil #4 Prod.Benzo[h]nhtracene <td>LPG Production</td> <td>Halogenated hydrocarbons (unspecified)</td> <td>secondary</td> <td>2.52e-18</td> <td>1.28e-19</td>	LPG Production	Halogenated hydrocarbons (unspecified)	secondary	2.52e-18	1.28e-19
CRT IncincrationHalogenated hydrocarbons (unspecified)secondary-4.35e-17Fuel Oil #4 Prod.Halogenated nutre (roganic)secondary-7.15e-16-3.62e-17Natural Gas Prod.HALON-1301secondary-1.18e-15-2.61e-16Natural Gas Prod.Mercury compoundssecondary-7.16e-15-3.63e-16Natural Gas Prod.Lead empdssecondary-1.18e-14-5.99e-10Natural Gas Prod.Lead empdssecondary-1.41e-14-7.13e-16CRT IncinerationHalogenated matter (organic)secondary-1.84e-14-9.30e-16Natural Gas Prod.Anthracenesecondary-4.79e-14-4.05e-15CRT IncinerationHALON-1301secondary-7.99e-14-4.05e-15Fuel Oil #4 Prod.Anthracenesecondary-4.84e-14-4.26e-15Natural Gas Prod.Schethyl chrysenesecondary-1.28e-13-1.11e-14Fuel Oil #4 Prod.Anthracenesecondary-2.28e-13-1.18e-14CRT IncinerationNickel empdssecondary-2.8ee-13-1.45e-14CRT IncinerationDiberzo[ah]anthracenesecondary-3.37e-14-3.22e-14Fuel Oil #4 Prod.Halogenated matter (organic)secondary-2.8ee-13-3.37e-14Natural Gas Prod.Shethyl chrysenesecondary-3.24e-14-2.38e-13-1.14e-14Fuel Oil #4 Prod.Benzo[b]iunanthenesecondary-4.69e-13-3.27e-14Fuel Oil #4 Prod.Benzo[b]iunanthenesecondary-4.69e	Natural Gas Prod.	Halogenated hydrocarbons (unspecified)	secondary	-2.95e-18	-1.50e-19
Fuel Oil #4 Prod. Halogenated hydrocarbons (unspecified) secondary -7, 15e-16 -3, 362e-17 Natural Gas Prod. Halogenated matter (organic) secondary -1, 18e-15 -5, 99e-17 Natural Gas Prod. HaLON-1301 secondary -7, 15e-16 -3, 362e-16 Natural Gas Prod. Lead cmpds secondary -1, 18e-14 -5, 99e-16 Natural Gas Prod. Nickel cmpds secondary -1, 18e-14 -5, 99e-16 Natural Gas Prod. Kel cmpds secondary -1, 18e-14 -2, 93oe-16 Natural Gas Prod. Anthracene secondary -1, 41e-14 -2, 33oe-15 CRT Incineration HALON-1301 secondary -2, 48e-13 -6, 49e-15 CRT Incineration Nickel cmpds secondary -2, 18e-13 -6, 49e-15 CRT Incineration Nickel cmpds secondary -2, 28e-13 -1, 47e-14 Fuel Oil #4 Prod. S-Methyl chrysene secondary -2, 28e-13 -1, 47e-14 Fuel Oil #4 Prod. Shethyl chrysene secondary -2, 28e-13 -1, 47e-14	CRT Incineration	Halogenated hydrocarbons (unspecified)	secondary	-4.35e-17	-2.20e-18
Natural Gas Prod. Halogenated matter (organic) secondary -1.18-15 5.99-17 Natural Gas Prod. HALON-1301 secondary -7.16-15 -3.63-16 Natural Gas Prod. Lead empds secondary -7.16-15 -3.63-16 Natural Gas Prod. Nickel empds secondary -1.41e-14 -5.99e-16 Natural Gas Prod. Nickel empds secondary -1.41e-14 -7.13e-16 Natural Gas Prod. Anthracene secondary -1.48e-14 -9.30e-16 Natural Gas Prod. Anthracene secondary -4.79e-14 -2.43e-15 CRT Incineration HALON-1301 secondary -8.41e-14 -4.26e-15 Natural Gas Prod. 5-Methyl chrysene secondary -2.18e-13 -6.49e-15 CRT Incineration Nickel empds secondary -2.18e-13 -1.11e-14 CRT Incineration Dibenzo[a,h]anthracene secondary -2.8e-13 -1.11e-14 CRT Incineration Dibenzo[a,h]anthracene secondary -5.29e-13 -2.28e-13 Autural Gas Prod.	Fuel Oil #4 Prod.	Halogenated hydrocarbons (unspecified)	secondary	-7.15e-16	-3.62e-17
Natural Gas Prod.HALON-1301secondary-5.15e-15-2.61e-16Natural Gas Prod.Mercury compoundssecondary-7.16e-15-3.63e-16Natural Gas Prod.Nickel empdssecondary-1.18e-14-5.79pe-16Natural Gas Prod.Halogenated matter (organic)secondary-1.41e-14-7.13e-16CRT IncinerationHALON-1301secondary-4.79e-14-2.43e-15CRT IncinerationHALON-1301secondary-7.99e-14-4.05e-15Natural Gas Prod.Anthracenesecondary-2.18e-13-6.49e-15CRT IncinerationNickel empdssecondary-2.18e-13-6.49e-15Natural Gas Prod.Shethyl chrysenesecondary-2.18e-13-1.11e-14Fuel Oil #4 Prod.Halogenated matter (organic)secondary-2.28e-13-1.41e-14CRT IncinerationDibenzo[a,h]amthracenesecondary-2.28e-13-2.28e-14CRT IncinerationDibenzo[a,h]amthracenesecondary-5.29e-13-2.68e-14Natural Gas Prod.Benzo[b,jk]fluoranthenesecondary-5.37e-14Natural Gas Prod.Fluorenesecondary-6.65e-13-3.37e-14Natural Gas Prod.Fluorenesecondary-7.34e-13-3.72e-14Natural Gas Prod.Fluorenesecondary-1.28e-13-4.16e-14Natural Gas Prod.Pyrenesecondary-1.28e-13-3.17e-14Natural Gas Prod.Pyrenesecondary-1.28e-12-5.28e-14Natural Gas Prod.Pyrene	Natural Gas Prod.	Halogenated matter (organic)	secondary	-1.18e-15	-5.99e-17
Natural Gas Prod.Mercury compoundssecondary-7.16e-15-3.63e-16Natural Gas Prod.Lead mpdssecondary-1.18e-14-5.99e-16CRT IncinerationHalogenated matter (organic)secondary-1.41e-14-7.13e-16CRT IncinerationHalogenated matter (organic)secondary-1.84e-14-9.30e-16Natural Gas Prod.Anthracenesecondary-7.99e-14-4.45e-15Fuel Oil #4 Prod.Anthracenesecondary-7.99e-14-4.45e-15Statter IG as Prod.S-Methyl chrysenesecondary-2.84e-13-6.49e-15CRT IncinerationNickel cmpdssecondary-2.86e-13-1.11e-14Fuel Oil #4 Prod.Halogenated matter (organic)secondary-2.86e-13-1.15e-14Fuel Oil #4 Prod.Halogenated matter (organic)secondary-4.49e-13-2.28e-14Fuel Oil #4 Prod.Benzolp,1,11urronesecondary-6.66e-13-3.34e-14Fuel Oil #4 Prod.Benzolp,1,11urronthenesecondary-6.66e-13-3.37e-14Natural Gas Prod.Fluoranthenesecondary-7.34e-13-3.22e-14Natural Gas Prod.Pitorenesecondary-1.24e-12-5.28e-14Natural Gas Prod.Dibenzolp,1,1nthracenesecondary-1.24e-12-5.28e-14Natural Gas Prod.Pitorenesecondary-1.24e-12-5.28e-14Natural Gas Prod.Dibenzolp,1,1nthracenesecondary-1.24e-12-5.28e-14Natural Gas Prod.Dibenzolp,1,1nthracenesecondary	Natural Gas Prod.	HALON-1301	secondary	-5.15e-15	-2.61e-16
Natural Gas Prod.Lead empdssecondary-1.18e-14-5.99e-16Natural Gas Prod.Nickel empdssecondary.1.41e-14-7.13e-16Natural Gas Prod.Anthracenesecondary.1.84e-14-9.30e-16Natural Gas Prod.Anthracenesecondary.7.99e-14.4.05e-15Fuel Oil #4 Prod.Anthracenesecondary.7.28e-13.6.49e-15Statural Gas Prod.S-Methyl chrysenesecondary.2.18e-13.1.11e-14Fuel Oil #4 Prod.Halogenated matter (organic)secondary.2.18e-13.1.11e-14Fuel Oil #4 Prod.Halogenated matter (organic)secondary.2.18e-13.1.11e-14Fuel Oil #4 Prod.Halogenated matter (organic)secondary.2.28e-14.2.28e-14Fuel Oil #4 Prod.S-Methyl chrysenesecondary.5.29e-13.2.68e-14Fuel Oil #4 Prod.Benzo[b,jk]fluoranthenesecondary.6.66e-13.3.24e-14Fuel Oil #4 Prod.Benzo[b,jk]fluoranthenesecondary.7.34e-13.3.72e-14Natural Gas Prod.Fluoranthenesecondary.7.34e-13.3.72e-14Natural Gas Prod.Fluoranthenesecondary.1.28e-12.6.73e-14Natural Gas Prod.Fluoranthenesecondary.1.24e-12.6.22e-14Vatural Gas Prod.Pyrenesecondary.1.24e-12.6.73e-14Natural Gas Prod.Opper (+1.&+2)secondary.1.24e-12.6.73e-14Natural Gas Prod.Copper (+1.&+2)secondary.1.24e-12.6.73	Natural Gas Prod.	Mercury compounds	secondary	-7.16e-15	-3.63e-16
Natural Gas Prod.Nickel empdssecondary-1.41e-14-7.13e-16CRT IncinerationHalogenated matter (organic)secondary-1.84e-14-9.30e-16Natural Gas Prod.Anthracenesecondary-4.79e-14-2.43e-15Fuel Oil 44 Prod.Anthracenesecondary-7.99e-144.05e-15Fuel Oil 44 Prod.Anthracenesecondary-1.28e-13-6.49e-15CRT IncinerationNickel empdssecondary-2.18e-13-1.11e-14Fuel Oil 44 Prod.Halogenated matter (organic)secondary-2.86e-13-1.15e-14Fuel Oil 44 Prod.Halogenated matter (organic)secondary-2.86e-13-1.45e-14Fuel Oil 44 Prod.Dibenzo[a,h]anthracenesecondary-4.49e-13-2.28e-14Fuel Oil 44 Prod.S-Methyl chrysenesecondary-6.40e-13-3.24e-14Fuel Oil 44 Prod.Benzo[b,h]flooranthenesecondary-6.65e-13-3.37e-14Natural Gas Prod.Benzo[b,h]flooranthenesecondary-7.34e-13-3.27e-14Natural Gas Prod.Fluorenesecondary-1.28e-12-5.28e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.28e-12-6.25e-14Natural Gas Prod.Pyrenesecondary-1.28e-12-5.28e-14Natural Gas Prod.Opper (+1 & ±2)secondary-1.28e-12-5.28e-14Natural Gas Prod.Opper (+1 & ±2)secondary-1.28e-12-7.50e-14Natural Gas Prod.Aromatic hydrocarbonssecondary-1.28e	Natural Gas Prod.	Lead cmpds	secondary	-1.18e-14	-5.99e-16
CRT IncinerationHalogenated matter (organic)secondary-1.84e-14-9.30e-16Natural Gas Prod.Anthracenesecondary-4.79e-14-2.43e-15CRT IncinerationHALON-1301secondary-7.99e-14-4.05e-15Natural Gas Prod.5-Methyl chrysenesecondary-1.28e-13-6.49e-15CRT IncinerationNickel cmpdssecondary-2.18e-13-1.11e-14Puel Oil #4 Prod.Halogenated matter (organic)secondary-2.28e-13-1.11e-14CRT IncinerationDibenzo[a,h]anthracenesecondary-2.28e-13-3.28e-14Fuel Oil #4 Prod.S-Methyl chrysenesecondary-5.29e-13-2.28e-14Fuel Oil #4 Prod.Benzo[b,j,k]fluoranthenesecondary-6.40e-13-3.24e-14CRT IncinerationBenzo[b,j,k]fluoranthenesecondary-7.34e-13-3.27e-14Natural Gas Prod.Fluoranthenesecondary-7.34e-13-3.27e-14Natural Gas Prod.Fluoranthenesecondary-1.28e-12-6.30e-13Vell Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-1.28e-12-6.30e-14Natural Gas Prod.Pyrenesecondary-1.28e-12-6.30e-14Natural Gas Prod.Pyrenesecondary-1.28e-12-6.30e-14Natural Gas Prod.Opper (+1 & ±2)secondary-1.28e-12-6.30e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.28e-12-7.30e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.28e-	Natural Gas Prod.	Nickel cmpds	secondary	-1.41e-14	-7.13e-16
Natural Gas Prod.Anthracenesecondary-4.79e-14-2.43e-15CRT IncinerationHALON-1301secondary-7.99e-14-4.05e-15Fuel Oil #4 Prod.Anthracenesecondary-8.41e-14-4.26e-15Mattral Gas Prod.5-Methyl chrysenesecondary-2.18e-13-6.49e-15CRT IncinerationNickel empdssecondary-2.18e-13-1.11e-14Fuel Oil #4 Prod.Halogenated matter (organic)secondary-2.28e-13-1.45e-14CRT IncinerationDibenzo[a,h]anthracenesecondary-2.28e-13-2.28e-14Luel Oil #4 Prod.5-Methyl chrysenesecondary-5.29e-13-2.28e-14Natural Gas Prod.Benzo[bj.j,k]fluoranthenesecondary-6.40e-13-3.27e-14Natural Gas Prod.Fluoranthenesecondary-6.40e-13-3.37e-14Natural Gas Prod.Fluorenthenesecondary-1.28e-12-5.28e-13Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-1.28e-12-5.28e-14Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-1.28e-12-5.28e-14Veul Oil #4 Prod.HALON-1301secondary-1.28e-12-5.28e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.28e-12-7.19e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.28e-12-7.19e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.28e-12-7.19e-14Natural Gas Prod.Benzo[g,h,i]perylenesecondary-1.42e-12 </td <td>CRT Incineration</td> <td>Halogenated matter (organic)</td> <td>secondary</td> <td>-1.84e-14</td> <td>-9.30e-16</td>	CRT Incineration	Halogenated matter (organic)	secondary	-1.84e-14	-9.30e-16
CRT IncinerationHALON-1301secondary-7.99e-14-4.05e-15Fuel Oil #4 Prod.Anthracenesecondary-8.41e-14-4.26e-15Natural Gas Prod.5-Methyl chrysenesecondary-1.28e-13-6.49e-15CRT IncinerationNickel empdssecondary-2.86e-13-1.11e-14Fuel Oil #4 Prod.Halogenated matter (organic)secondary-2.86e-13-1.45e-14CRT IncinerationDibenzo[a,b]anthracenesecondary-5.29e-13-2.28e-14Fuel Oil #4 Prod.5-Methyl chrysenesecondary-6.65e-13-3.37e-14Natural Gas Prod.Benzo[b,j,k]fluoranthenesecondary-6.65e-13-3.37e-14Natural Gas Prod.Fluorenesecondary-7.34e-13-3.72e-14Natural Gas Prod.Fluorenesecondary-1.23e-12-6.22e-13Heul Oil #4 Prod.Dibenzo[a,b]anthracenesecondary-1.23e-12-6.22e-14Fuel Oil #4 Prod.HALON-1301secondary-1.23e-12-6.32e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.33e-12-6.73e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.32e-12-7.20e-14Natural Gas Prod.Dibenzo[a,b]anthracenesecondary-1.24e-12-7.30e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.24e-12-7.19e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-1.42e-12-7.20e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-1.42e-12 <t< td=""><td>Natural Gas Prod.</td><td>Anthracene</td><td>secondary</td><td>-4.79e-14</td><td>-2.43e-15</td></t<>	Natural Gas Prod.	Anthracene	secondary	-4.79e-14	-2.43e-15
Fuel Oil #4 Prod.Anthracenesecondary-8.41e-14-4.26e-15Natural Gas Prod.5-Methyl chrysenesecondary-1.28e-13-6.49e-15CRT IncinerationNickel emplssecondary-2.86e-13-1.11e-14Puel Oil #4 Prod.Halogenated matter (organic)secondary-2.86e-13-1.45e-14CRT IncinerationDibenzo[a,h]anthracenesecondary-2.86e-13-2.28e-14Fuel Oil #4 Prod.5-Methyl chrysenesecondary-6.40e-13-2.28e-14Natural Gas Prod.Benzo[b,jk]fluoranthenesecondary-6.65e-13-3.37e-14Natural Gas Prod.Fluoranthenesecondary-6.65e-13-3.37e-14Natural Gas Prod.Fluorenesecondary-7.34e-13-3.22e-14Natural Gas Prod.Pluorenesecondary-1.23e-12-6.22e-14Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-1.23e-12-6.22e-14Fuel Oil #4 Prod.HALON-1301secondary-1.24e-12-6.30e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.42e-12-7.30e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.30e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.28e-12-7.30e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.28e-12-7.30e-14Natural Gas Prod.Benzo[b,l]nutracenesecondary-1.42e-12-7.30e-14Natural Gas Prod.Benzo[b,l]nutratenesecondary-1.42e-12<	CRT Incineration	HALON-1301	secondary	-7.99e-14	-4.05e-15
Natural Gas Prod.5-Methyl chrysenesecondary-1.28e-13-6.49e-15CRT IncinerationNickel empdssecondary-2.18e-13-1.11e-14Fuel Oil #4 Prod.Halogenated matter (organic)secondary-2.86e-13-1.45e-14CRT IncinerationDibenzo[a,h]anthracenesecondary-2.26e-13-2.26e-14Natural Gas Prod.S-Methyl chrysenesecondary-6.40e-13-3.24e-14CRT IncinerationBenzo[b,j,k]fluoranthenesecondary-6.65e-13-3.37e-14Natural Gas Prod.Fluoranthenesecondary-7.34e-13-3.72e-14Natural Gas Prod.Fluoranthenesecondary-8.22e-13-4.16e-14Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-8.22e-13-4.16e-14Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-1.23e-12-6.22e-14Fuel Oil #4 Prod.HALON-1301secondary-1.23e-12-6.30e-14Natural Gas Prod.Pyrenesecondary-1.42e-12-7.9e-14Natural Gas Prod.Aromatic hydrocarbonssecondary-1.42e-12-7.9e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.9e-14Natural Gas Prod.Aromatic hydrocarbonssecondary-1.42e-12-7.9e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.9e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-1.42e-12-7.9e-14Natural Gas Prod.Benzo[a,h]aptrenesecondary </td <td>Fuel Oil #4 Prod.</td> <td>Anthracene</td> <td>secondary</td> <td>-8.41e-14</td> <td>-4.26e-15</td>	Fuel Oil #4 Prod.	Anthracene	secondary	-8.41e-14	-4.26e-15
CRT IncinerationNickel empdssecondary-2.18e-13-1.11e-14Fuel Oil #4 Prod.Halogenated matter (organic)secondary-2.8ee-13-1.45e-14CRT IncinerationDibenzo[a,h]anthracenesecondary-4.49e-13-2.28e-14Fuel Oil #4 Prod.5-Methyl chrysenesecondary-5.29e-13-2.28e-14Rutural Gas Prod.Benzo[b,j,k]fluoranthenesecondary-6.65e-13-3.37e-14Natural Gas Prod.Fluoranthenesecondary-6.65e-13-3.37e-14Natural Gas Prod.Fluoranthenesecondary-7.34e-13-3.72e-14Natural Gas Prod.Fluorenesecondary-1.04e-12-5.28e-14Vatural Gas Prod.Pyrenesecondary-1.23e-12-6.22e-14Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-1.24e-12-6.30e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.42e-12-7.19e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.42e-12-7.19e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.19e-14Natural Gas Prod.Benzo[a,h]anthracenesecondary-1.42e-12-7.50e-14Natural Gas Prod.Benzo[a,h]anthracenesecondary-1.42e-12-7.50e-14Natural Gas Prod.Benzo[a,h]anthracenesecondary-1.42e-12-7.50e-14Natural Gas Prod.Benzo[a,h]anthracenesecondary-1.42e-12-7.50e-14Natural Gas Prod.Benzo[a,h]aperylenesecondary<	Natural Gas Prod.	5-Methyl chrysene	secondary	-1.28e-13	-6.49e-15
Fuel Oil #4 Prod.Halogenated matter (organic)secondary-2.86e-13-1.45e-14CRT IncinerationDibenzo[a,h]anthracenesecondary-4.49e-13-2.28e-14Fuel Oil #4 Prod.5-Methyl chrysnensecondary-5.29e-13-2.68e-14Natural Gas Prod.Benzo[b,j,k]fluoranthenesecondary-6.65e-13-3.34e-14CRT IncinerationBenzo[b,jhluoranthenesecondary-6.65e-13-3.37e-14Natural Gas Prod.Fluoranthenesecondary-7.34e-13-3.72e-14Natural Gas Prod.Fluoranthenesecondary-1.04e-12-5.28e-14Vell 0il #4 Prod.Dibenzo[a,h]anthracenesecondary-1.23e-12-6.22e-14Vuel Oil #4 Prod.HALON-1301secondary-1.24e-12-5.30e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.42e-12-7.19e-14Natural Gas Prod.Obenzo[a,h]anthracenesecondary-1.42e-12-7.20e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.20e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.20e-14Fuel Oil #4 Prod.Benzo[a,h]anthracenesecondary-1.42e-12-8.80e-14Natural Gas Prod.Benzo[a,h]anthracenesecondary-1.48e-12-7.20e-14Fuel Oil #4 Prod.Benzo[a,h]anthracenesecondary-1.48e-12-8.80e-14Natural Gas Prod.Benzo[a,h]anthracenesecondary-1.48e-12-8.80e-14Natural Gas Prod.Benzo[a,h]p	CRT Incineration	Nickel cmpds	secondary	-2.18e-13	-1.11e-14
CRT IncinerationDibenzo[a,h]anthracenesecondary-4.49e-13-2.28e-14Fuel Oil #4 Prod.5-Methyl chrysenesecondary-5.29e-13-2.68e-14Natural Gas Prod.Benzo[b,j,k]fluoranthenesecondary-6.40e-13-3.24e-14CRT IncinerationBenzo[b]fluoranthenesecondary-6.65e-13-3.37e-14Natural Gas Prod.Fluorenesecondary-7.34e-13-3.72e-14Natural Gas Prod.Fluorenesecondary-1.04e-12-5.28e-14Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-1.23e-12-6.22e-14Fuel Oil #4 Prod.HALON-1301secondary-1.24e-12-6.30e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.32e-12-6.30e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.42e-12-7.19e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.20e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.20e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-1.42e-12-7.20e-14Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.48e-12-7.20e-14Natural Gas Prod.Benzo[a,h]perylenesecondary-1.88e-12-8.00e-14Natural Gas Prod.Benzo[a,h]perylenesecondary-1.42e-12-7.20e-14Fuel Oil #4 Prod.Benzo[a,h]perylenesecondary-1.74e-12-8.11e-14Fuel Oil #4 Prod.Benzo[a,h]perylenesecondary <td>Fuel Oil #4 Prod.</td> <td>Halogenated matter (organic)</td> <td>secondary</td> <td>-2.86e-13</td> <td>-1.45e-14</td>	Fuel Oil #4 Prod.	Halogenated matter (organic)	secondary	-2.86e-13	-1.45e-14
Fuel Oil #4 Prod.5-Methyl chrysenesecondary-5.29e-13-2.68e-14Natural Gas Prod.Benzo[b,j,k]fluoranthenesecondary-6.40e-13-3.24e-14CRT IncinerationBenzo[b]fluoranthenesecondary-6.65e-13-3.37e-14Natural Gas Prod.Fluoranthenesecondary-7.34e-13-3.72e-14Natural Gas Prod.Fluoranthenesecondary-8.22e-13-4.16e-14Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-1.04e-12-5.28e-14Natural Gas Prod.Pyrenesecondary-1.24e-12-6.30e-14Fuel Oil #4 Prod.HALON-1301secondary-1.24e-12-6.30e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.33e-12-6.73e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.42e-12-7.10e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.50e-14Natural Gas Prod.Benzo[a,h]anthracenesecondary-1.48e-12-7.50e-14Natural Gas Prod.Benzo[a,h]anthracenesecondary-1.48e-12-7.50e-14Natural Gas Prod.Benzo[a,h]aprenesecondary-1.48e-12-7.50e-14Natural Gas Prod.Benzo[a,h]iperylenesecondary-1.60e-12-8.11e-14Fuel Oil #4 Prod.Benzo[a,h]iperylenesecondary-1.73e-12-8.77e-14Fuel Oil #4 Prod.Benzo[a,h]iperylenesecondary-1.72e-12-9.52e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.78e-12<	CRT Incineration	Dibenzo[a,h]anthracene	secondary	-4.49e-13	-2.28e-14
Natural Gas Prod.Benzo[b,j,k]fluoranthenesecondary-6.40e-13-3.24e-14CRT IncinerationBenzo[b]fluoranthenesecondary-6.65e-13-3.37e-14Natural Gas Prod.Fluoranthenesecondary-7.34e-13-3.72e-14Natural Gas Prod.Fluorenesecondary-8.22e-13-4.16e-14Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-1.23e-12-6.22e-14Natural Gas Prod.Pyrenesecondary-1.23e-12-6.22e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.32e-12-6.73e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.42e-12-7.19e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.19e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.19e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-1.42e-12-7.50e-14Natural Gas Prod.Benzo[a,h]anthracenesecondary-1.58e-12-8.00e-14Fuel Oil #4 Prod.Benzo[a,h]preylenesecondary-1.74e-12-8.17e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-8.17e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-8.17e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-8.17e-14Fuel Oil #4 Prod.ProdFluoranthenesecondary-1.74e-12-8.17e-14Fuel Oil #4 Prod.ProdProdenthene <td>Fuel Oil #4 Prod.</td> <td>5-Methyl chrysene</td> <td>secondary</td> <td>-5.29e-13</td> <td>-2.68e-14</td>	Fuel Oil #4 Prod.	5-Methyl chrysene	secondary	-5.29e-13	-2.68e-14
CRT IncinerationBenzo[b]fluoranthenesecondary-6.65e-13-3.37e-14Natural Gas Prod.Fluoranthenesecondary-7.34e-13-3.72e-14Natural Gas Prod.Fluorenesecondary-8.22e-13-4.16e-14Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-1.28e-12-6.22e-14Natural Gas Prod.Pyrenesecondary-1.24e-12-6.30e-14Fuel Oil #4 Prod.HALON-1301secondary-1.24e-12-6.30e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.33e-12-6.73e-14Natural Gas Prod.Aromatic hydrocarbonssecondary-1.42e-12-7.19e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-1.42e-12-7.20e-14Veul Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.58e-12-8.00e-14Natural Gas Prod.Benzo[g,h,i]perylenesecondary-1.58e-12-8.00e-14Natural Gas Prod.Benzo[g,h,i]perylenesecondary-1.58e-12-8.00e-14Natural Gas Prod.Benzo[g,h,i]perylenesecondary-1.73e-12-8.77e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-8.81e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-9.70e-14Vatural Gas Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-8.81e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.74e-12-8.81e-14Fuel Oil #4 Prod.Pyrenesecondary-1.74e-12-	Natural Gas Prod.	Benzo[b,j,k]fluoranthene	secondary	-6.40e-13	-3.24e-14
Natural Gas Prod.Fluoranthenesecondary-7.34e-13-3.72e-14Natural Gas Prod.Fluorenesecondary-8.22e-13-4.16e-14Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-1.04e-12-5.28e-14Natural Gas Prod.Pyrenesecondary-1.23e-12-6.22e-14Fuel Oil #4 Prod.HALON-1301secondary-1.24e-12-6.30e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.42e-12-7.19e-14Natural Gas Prod.Aromatic hydrocarbonssecondary-1.42e-12-7.19e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.20e-14Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.58e-12-8.00e-14Natural Gas Prod.Benzo[a,h]gprenesecondary-1.58e-12-8.00e-14Natural Gas Prod.Benzo[a,h]gprenesecondary-1.58e-12-8.00e-14Vatural Gas Prod.Benzo[a,h]gprelenesecondary-1.58e-12-8.10e-14Fuel Oil #4 Prod.Benzo[a,h]jperylenesecondary-1.58e-12-8.10e-14Fuel Oil #4 Prod.Benzo[b,h]jperylenesecondary-1.60e-12-8.11e-14Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.73e-12-8.77e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.78e-12-9.52e-14Fuel Oil #4 Prod.Pyrenesecondary-1.60e-12-8.11e-14Fuel Oil #4 Prod.Pyrenesecondary-2.09e-12-1.06e-13<	CRT Incineration	Benzo[b]fluoranthene	secondary	-6.65e-13	-3.37e-14
Natural Gas Prod.Fluorenesecondary-8.22e-13-4.16e-14Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-1.04e-12-5.28e-14Natural Gas Prod.Pyrenesecondary-1.23e-12-6.22e-14Fuel Oil #4 Prod.HALON-1301secondary-1.24e-12-6.30e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.32e-12-7.19e-14Natural Gas Prod.Aromatic hydrocarbonssecondary-1.42e-12-7.19e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.20e-14Fuel Oil #4 Prod.Benzo[a,h]anthracenesecondary-1.48e-12-7.50e-14Natural Gas Prod.Benzo[a,h]aprenesecondary-1.58e-12-8.00e-14Natural Gas Prod.Benzo[a,h]prenesecondary-1.60e-12-8.11e-14Fuel Oil #4 Prod.Benzo[a,h]preylenesecondary-1.73e-12-8.77e-14Fuel Oil #4 Prod.Benzo[a,h]preylenesecondary-1.73e-12-8.77e-14Fuel Oil #4 Prod.Benzo[a,h]preylenesecondary-1.74e-12-8.81e-14Fuel Oil #4 Prod.Benzo[a,h]iperylenesecondary-1.74e-12-8.81e-14Fuel Oil #4 Prod.Pyrenesecondary-1.74e-12-9.70e-14Fuel Oil #4 Prod.Pyrenesecondary-1.91e-12-9.70e-14Fuel Oil #4 Prod.Pyrenesecondary-2.26e-12-1.16e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.26e-12-1.16e-13 <td< td=""><td>Natural Gas Prod.</td><td>Fluoranthene</td><td>secondary</td><td>-7.34e-13</td><td>-3.72e-14</td></td<>	Natural Gas Prod.	Fluoranthene	secondary	-7.34e-13	-3.72e-14
Fuel Oil #4 Prod.Dibenzo[a,h]anthracenesecondary-1.04e-12-5.28e-14Natural Gas Prod.Pyrenesecondary-1.23e-12-6.22e-14Fuel Oil #4 Prod.HALON-1301secondary-1.24e-12-6.30e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.33e-12-6.73e-14Natural Gas Prod.Aromatic hydrocarbonssecondary-1.42e-12-7.19e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.20e-14Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.48e-12-7.50e-14Natural Gas Prod.Benzo[g,h,i]perylenesecondary-1.58e-12-8.00e-14Natural Gas Prod.Benzo[g,h,i]perylenesecondary-1.73e-12-8.77e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.73e-12-8.77e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.78e-12-9.52e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-8.81e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-8.81e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.91e-12-9.70e-14Fuel Oil #4 Prod.Pyrenesecondary-1.24e-12-7.00e-13Fuel Oil #4 Prod.Pyrenesecondary-2.20e-12-1.16e-13Fuel Oil #4 Prod.Pyrenesecondary-2.20e-12-1.16e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.26e-12-1.14e	Natural Gas Prod.	Fluorene	secondary	-8.22e-13	-4.16e-14
Natural Gas Prod.Pyrenesecondary-1.23e-12-6.22e-14Fuel Oil #4 Prod.HALON-1301secondary-1.24e-12-6.30e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.33e-12-6.73e-14Natural Gas Prod.Aromatic hydrocarbonssecondary-1.42e-12-7.19e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.20e-14Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.48e-12-7.50e-14Natural Gas Prod.Benzo[a,h]aptracenesecondary-1.58e-12-8.00e-14Natural Gas Prod.Benzo[g,h,i]perylenesecondary-1.60e-12-8.11e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-8.77e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-9.52e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.91e-12-9.70e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-1.91e-12-9.70e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.09e-12-1.06e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.26e-12-1.14e-13Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-2.26e-12-1.14e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.32e-12-1.14e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.32e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary<	Fuel Oil #4 Prod.	Dibenzo[a,h]anthracene	secondary	-1.04e-12	-5.28e-14
Fuel Oil #4 Prod.HALON-1301secondary-1.24e-12-6.30e-14Natural Gas Prod.Copper (+1 & +2)secondary-1.33e-12-6.73e-14Natural Gas Prod.Aromatic hydrocarbonssecondary-1.42e-12-7.19e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.20e-14Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.48e-12-7.50e-14Natural Gas Prod.Benzo[a]pyrenesecondary-1.58e-12-8.00e-14Natural Gas Prod.Benzo[g,h,i]perylenesecondary-1.73e-12-8.11e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.73e-12-8.77e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-8.81e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.91e-12-9.52e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.91e-12-9.52e-14Fuel Oil #4 Prod.Pyrenesecondary-1.91e-12-9.52e-14Fuel Oil #4 Prod.Pyrenesecondary-1.91e-12-9.52e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.26e-12-1.16e-13Natural Gas Prod.Acenaphthenesecondary-2.26e-12-1.14e-13Natural Gas Prod.Fluorenesecondary-2.32e-12-1.18e-13Natural Gas Prod.Fluorenesecondary-2.32e-12-1.18e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.18e-13Fue	Natural Gas Prod.	Pyrene	secondary	-1.23e-12	-6.22e-14
Natural Gas Prod.Copper (+1 & +2)secondary-1.33e-12-6.73e-14Natural Gas Prod.Aromatic hydrocarbonssecondary-1.42e-12-7.19e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.20e-14Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.48e-12-7.50e-14Natural Gas Prod.Benzo[a]pyrenesecondary-1.58e-12-8.00e-14Natural Gas Prod.Benzo[g,h,i]perylenesecondary-1.60e-12-8.11e-14Fuel Oil #4 Prod.Mercury compoundssecondary-1.73e-12-8.77e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-128.81e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.88e-12-9.52e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.91e-12-9.70e-14Fuel Oil #4 Prod.Pyrenesecondary-1.91e-12-9.70e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.91e-12-9.70e-14Fuel Oil #4 Prod.Pyrenesecondary-1.91e-12-9.70e-14Fuel Oil #4 Prod.Pyrenesecondary-1.91e-12-9.70e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.09e-12-1.06e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.26e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.18e-13Natural Gas Prod.Biphenylsecondary-2.35e-12-1.19e-13Fuel Oi	Fuel Oil #4 Prod.	HALON-1301	secondary	-1.24e-12	-6.30e-14
Natural Gas Prod.Aromatic hydrocarbonssecondary-1.42e-12-7.19e-14Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.20e-14Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.48e-12-7.50e-14Natural Gas Prod.Benzo[a]pyrenesecondary-1.58e-12-8.00e-14Natural Gas Prod.Benzo[g,h,i]perylenesecondary-1.60e-12-8.11e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.73e-12-8.77e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-8.81e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.91e-12-9.52e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.91e-12-9.70e-14Fuel Oil #4 Prod.Pyrenesecondary-1.91e-12-9.70e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.91e-12-9.70e-14Fuel Oil #4 Prod.Pyrenesecondary-1.91e-12-9.70e-14Fuel Oil #4 Prod.Pyrenesecondary-1.91e-12-9.70e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.20e-12-1.10e-13Natural Gas Prod.Acenaphthenesecondary-2.22e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.14e-13Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-2.32e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.14e-13Fuel Oi	Natural Gas Prod.	Copper (+1 & +2)	secondary	-1.33e-12	-6.73e-14
Natural Gas Prod.Dibenzo[a,h]anthracenesecondary-1.42e-12-7.20e-14Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.48e-12-7.50e-14Natural Gas Prod.Benzo[a]pyrenesecondary-1.58e-12-8.00e-14Natural Gas Prod.Benzo[g,h,i]perylenesecondary-1.60e-12-8.11e-14Fuel Oil #4 Prod.Mercury compoundssecondary-1.73e-12-8.77e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-8.81e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-9.52e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.91e-12-9.70e-14Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.91e-12-9.70e-14Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.26e-12-1.14e-13Fuel Oil #4 Prod.Pyrenesecondary-2.09e-12-1.06e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.26e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.14e-13Fuel Oil #4 Prod.Biphenylsecondary-2.32e-12-1.19e-13Fuel Oil #4 Prod.Biphenylsecondary-2.32e-12-1.19e-13Fuel Oi	Natural Gas Prod.	Aromatic hydrocarbons	secondary	-1.42e-12	-7.19e-14
Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.48e-12-7.50e-14Natural Gas Prod.Benzo[a]pyrenesecondary-1.58e-12-8.00e-14Natural Gas Prod.Benzo[g,h,i]perylenesecondary-1.60e-12-8.11e-14Fuel Oil #4 Prod.Mercury compoundssecondary-1.73e-12-8.77e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-8.81e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.91e-12-9.52e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.91e-12-9.70e-14Fuel Oil #4 Prod.Benzo[b]fluoranthenesecondary-1.91e-12-9.70e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.09e-12-1.06e-13Natural Gas Prod.Acenaphthenesecondary-2.26e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.18e-13Natural Gas Prod.Biphenylsecondary-2.32e-12-1.18e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.19e-13Fuel Oil #4 Prod.Biphenylsecondary-2.35e-12-1.19e-13Fuel Oil #4 Prod.Biphenylsecondary-2.35e-12-1.19e-13Fuel Oil #4 Prod.Biphenylsecondary-2.35e-12-1.19e-13Fuel Oil #4 Prod.Biphenylsecondary-2.36e-12-1.19e-13Fuel Oil #4 Prod.Benzo[a]pyrenesecondary-2.36e-12-1.19e-13Fuel Oil #4 Prod.Bip	Natural Gas Prod.	Dibenzo[a,h]anthracene	secondary	-1.42e-12	-7.20e-14
Natural Gas Prod.Benzo[a]pyrenesecondary-1.58e-12-8.00e-14Natural Gas Prod.Benzo[g,h,i]perylenesecondary-1.60e-12-8.11e-14Fuel Oil #4 Prod.Mercury compoundssecondary-1.73e-12-8.77e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-8.81e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.88e-12-9.52e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.91e-12-9.70e-14Fuel Oil #4 Prod.Pyrenesecondary-2.09e-12-1.06e-13Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.26e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.18e-13Natural Gas Prod.Biphenylsecondary-2.32e-12-1.18e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.18e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.19e-13Fuel Oil #4 Prod.Biphenylsecondary-2.35e-12-1.19e-13Fuel Oil #4 Prod.Biphenylsecondary-2.35e-12-1.19e-13Fuel Oil #4 Prod.Biphenylsecondary-2.35e-12-1.19e-13Fuel Oil #4 Prod.Benzo[a]pyrenesecondary-2.35e-12-1.19e-13Fuel Oil #4 Prod.Biphenylsecondary-2.35e-12-1.19e-13Fuel Oil #4 Prod.Benzo[a]pyrenesecondary-2.35e-12-1.19e-13	Fuel Oil #4 Prod.	Benzo[b]fluoranthene	secondary	-1.48e-12	-7.50e-14
Natural Gas Prod.Benzo[g,h,i]perylenesecondary-1.60e-12-8.11e-14Fuel Oil #4 Prod.Mercury compoundssecondary-1.73e-12-8.77e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-8.81e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.88e-12-9.52e-14Fuel Oil #4 Prod.Pyrenesecondary-1.91e-12-9.70e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.09e-12-1.06e-13Natural Gas Prod.Acenaphthenesecondary-2.26e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.18e-13Natural Gas Prod.Biphenylsecondary-2.32e-12-1.19e-13Fuel Oil #4 Prod.Biphenylsecondary-2.35e-12-1.19e-13Fuel Oil #4 Prod.Biphenylsecondary-2.35e-12-1.19e-13	Natural Gas Prod.	Benzo[a]pyrene	secondary	-1.58e-12	-8.00e-14
Fuel Oil #4 Prod.Mercury compoundssecondary-1.73e-12-8.77e-14Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-8.81e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.88e-12-9.52e-14Fuel Oil #4 Prod.Pyrenesecondary-1.91e-12-9.70e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.09e-12-1.06e-13Natural Gas Prod.Acenaphthenesecondary-2.26e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.18e-13Natural Gas Prod.Biphenylsecondary-2.35e-12-1.19e-13Fuel Oil #4 Prod.Benzo[alpyrenesecondary-2.35e-12-1.19e-13	Natural Gas Prod.	Benzo[g,h,i]pervlene	secondary	-1.60e-12	-8.11e-14
Fuel Oil #4 Prod.Benzo[g,h,i]perylenesecondary-1.74e-12-8.81e-14Fuel Oil #4 Prod.Fluoranthenesecondary-1.88e-12-9.52e-14Fuel Oil #4 Prod.Pyrenesecondary-1.91e-12-9.70e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.09e-12-1.06e-13Natural Gas Prod.Acenaphthenesecondary-2.26e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.18e-13Natural Gas Prod.Biphenylsecondary-2.32e-12-1.19e-13Fuel Oil #4 Prod.Biphenylsecondary-2.35e-12-1.19e-13Fuel Oil #4 Prod.Benzo[a]nyrenesecondary-2.36e-12-1.19e-13	Fuel Oil #4 Prod.	Mercury compounds	secondary	-1.73e-12	-8.77e-14
Fuel Oil #4 Prod.Fluoranthenesecondary-1.88e-12-9.52e-14Fuel Oil #4 Prod.Pyrenesecondary-1.91e-12-9.70e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.09e-12-1.06e-13Natural Gas Prod.Acenaphthenesecondary-2.26e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.18e-13Natural Gas Prod.Biphenylsecondary-2.35e-12-1.19e-13Fuel Oil #4 Prod.Benzo[a]pyrenesecondary-2.36e-12-1.19e-13	Fuel Oil #4 Prod.	Benzo[g,h,i]perylene	secondary	-1.74e-12	-8.81e-14
Fuel Oil #4 Prod.Pyrenesecondary-1.91e-12-9.70e-14Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.09e-12-1.06e-13Natural Gas Prod.Acenaphthenesecondary-2.26e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.18e-13Natural Gas Prod.Biphenylsecondary-2.35e-12-1.19e-13Fuel Oil #4 Prod.Benzo[a]pyrenesecondary-2.36e-12-1.19e-13	Fuel Oil #4 Prod.	Fluoranthene	secondary	-1.88e-12	-9.52e-14
Natural Gas Prod.Benzo[b]fluoranthenesecondary-2.09e-12-1.06e-13Natural Gas Prod.Acenaphthenesecondary-2.26e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.18e-13Natural Gas Prod.Bipheny1secondary-2.35e-12-1.19e-13Fuel Oil #4 ProdBenzo[a]nyrenesecondary-2.36e-12-1.19e-13	Fuel Oil #4 Prod.	Pyrene	secondary	-1.91e-12	-9.70e-14
Natural Gas Prod.Acenaphtenesecondary-2.26e-12-1.14e-13Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.18e-13Natural Gas Prod.Biphenylsecondary-2.35e-12-1.19e-13Fuel Oil #4 Prod.Benzolalpyrenesecondary-2.36e-12-1.19e-13	Natural Gas Prod.	Benzo[b]fluoranthene	secondary	-2.09e-12	-1.06e-13
Fuel Oil #4 Prod.Fluorenesecondary-2.32e-12-1.18e-13Natural Gas Prod.Biphenylsecondary-2.35e-12-1.19e-13Fuel Oil #4 ProdBenzo[a]pyrenesecondary-2.36e-12-1.19e-13	Natural Gas Prod.	Acenaphthene	secondary	-2.26e-12	-1.14e-13
Natural Gas Prod.Biphenylsecondary-2.35e-12-1.19e-13Fuel Oil #4 ProdBenzolalpyrenesecondary-2.36e-12-1.19e-13	Fuel Oil #4 Prod.	Fluorene	secondary	-2.32e-12	-1.18e-13
Fuel Oil #4 Prod Benzolalnyrene secondary _2 36e-12 _1 19e-13	Natural Gas Prod.	Biphenyl	secondary	-2.35e-12	-1.19e-13
-2.30C-12 -1.17C-15	Fuel Oil #4 Prod.	Benzo[a]pyrene	secondary	-2.36e-12	-1.19e-13

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Acetophenone	secondary	-2.46e-12	-1.24e-13
Natural Gas Prod.	Indeno(1,2,3-cd)pyrene	secondary	-2.47e-12	-1.25e-13
Fuel Oil #4 Prod.	Benzo[b,j,k]fluoranthene	secondary	-2.65e-12	-1.34e-13
Natural Gas Prod.	Chrysene	secondary	-2.71e-12	-1.37e-13
Natural Gas Prod.	Benzo[a]anthracene	secondary	-2.81e-12	-1.42e-13
Fuel Oil #4 Prod.	Lead cmpds	secondary	-2.86e-12	-1.45e-13
Fuel Oil #4 Prod.	Indeno(1,2,3-cd)pyrene	secondary	-3.00e-12	-1.52e-13
Fuel Oil #4 Prod.	Nickel cmpds	secondary	-3.40e-12	-1.72e-13
Natural Gas Prod.	Acenaphthylene	secondary	-3.51e-12	-1.78e-13
CRT Incineration	Anthracene	secondary	-3.73e-12	-1.89e-13
Fuel Oil #4 Prod.	Benzo[a]anthracene	secondary	-3.93e-12	-1.99e-13
Natural Gas Prod.	Cumene	secondary	-3.95e-12	-2.00e-13
Fuel Oil #4 Prod.	Chrysene	secondary	-3.96e-12	-2.01e-13
Natural Gas Prod.	Ethyl Chloride	secondary	-4.66e-12	-2.36e-13
Fuel Oil #4 Prod.	Acenaphthene	secondary	-5.79e-12	-2.93e-13
Natural Gas Prod.	1,1,1-Trichloroethane	secondary	-6.58e-12	-3.33e-13
Natural Gas Prod.	Ethylene dibromide	secondary	-6.98e-12	-3.54e-13
Natural Gas Prod.	Cadmium cmpds	secondary	-7.04e-12	-3.57e-13
Fuel Oil #4 Prod.	Acenaphthylene	secondary	-7.42e-12	-3.76e-13
CRT Incineration	2-Methylnaphthalene	secondary	-8.71e-12	-4.41e-13
Fuel Oil #4 Prod.	Biphenyl	secondary	-9.73e-12	-4.93e-13
Fuel Oil #4 Prod.	Acetophenone	secondary	-1.02e-11	-5.14e-13
Fuel Oil #4 Prod.	Chlorine	secondary	-1.58e-11	-7.99e-13
Fuel Oil #4 Prod.	Cumene	secondary	-1.63e-11	-8.26e-13
Natural Gas Prod.	Vinyl acetate	secondary	-1.73e-11	-8.75e-13
Natural Gas Prod.	Styrene	secondary	-1.77e-11	-8.96e-13
Fuel Oil #4 Prod.	2-Methylnaphthalene	secondary	-1.81e-11	-9.19e-13
Fuel Oil #4 Prod.	Ethyl Chloride	secondary	-1.93e-11	-9.77e-13
CRT Incineration	Chlorine	secondary	-1.94e-11	-9.81e-13
CRT Incineration	Copper (+1 & +2)	secondary	-2.06e-11	-1.04e-12
CRT Incineration	Aromatic hydrocarbons	secondary	-2.20e-11	-1.12e-12
CRT Incineration	Zinc (+2)	secondary	-2.24e-11	-1.13e-12
Natural Gas Prod.	Methyl tert-butyl ether	secondary	-2.42e-11	-1.23e-12
Natural Gas Prod.	2-Methylnaphthalene	secondary	-2.71e-11	-1.37e-12
CRT Incineration	3-Methylcholanthrene	secondary	-2.72e-11	-1.38e-12
Fuel Oil #4 Prod.	1,1,1-Trichloroethane	secondary	-2.72e-11	-1.38e-12
Fuel Oil #4 Prod.	Ethylene dibromide	secondary	-2.89e-11	-1.46e-12
Fuel Oil #4 Prod.	Zinc (+2)	secondary	-3.21e-11	-1.63e-12
CRT Incineration	5-Methyl chrysene	secondary	-3.27e-11	-1.66e-12
Natural Gas Prod.	Phenanthrene	secondary	-3.53e-11	-1.79e-12
CRT Incineration	Benzo[g,h,i]perylene	secondary	-4.06e-11	-2.06e-12
Natural Gas Prod.	Aluminum (elemental)	secondary	-4.26e-11	-2.16e-12
Natural Gas Prod.	Ethylbenzene	secondary	-4.85e-11	-2.45e-12
CRT Incineration	Dichlorobenzene (mixed isomers)	secondary	-4.90e-11	-2.48e-12
Fuel Oil #4 Prod.	3-Methylcholanthrene	secondary	-5.66e-11	-2.87e-12
CRT Incineration	Benzo[a]pyrene	secondary	-5.70e-11	-2.89e-12
Fuel Oil #4 Prod.	Vinyl acetate	secondary	-7.14e-11	-3.62e-12
Fuel Oil #4 Prod.	Styrene	secondary	-7.31e-11	-3.71e-12
CRT Incineration	Pyrene	secondary	-7.81e-11	-3.96e-12
Fuel Oil #4 Prod.	Phenanthrene	secondary	-7.87e-11	-3.99e-12

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Chromium (III)	secondary	-8.08e-11	-4.09e-12
Natural Gas Prod.	3-Methylcholanthrene	secondary	-8.46e-11	-4.29e-12
CRT Incineration	Indeno(1,2,3-cd)pyrene	secondary	-9.14e-11	-4.63e-12
Natural Gas Prod.	2,4-Dinitrotoluene	secondary	-9.70e-11	-4.91e-12
Fuel Oil #4 Prod.	Methyl tert-butyl ether	secondary	-1.00e-10	-5.08e-12
CRT Incineration	Fluoranthene	secondary	-1.01e-10	-5.10e-12
CRT Incineration	Benzo[a]anthracene	secondary	-1.20e-10	-6.06e-12
Natural Gas Prod.	Chlorobenzene	secondary	-1.22e-10	-6.17e-12
CRT Incineration	Fluorene	secondary	-1.29e-10	-6.53e-12
Natural Gas Prod.	Dichloromethane	secondary	-1.46e-10	-7.38e-12
CRT Incineration	Chrysene	secondary	-1.49e-10	-7.57e-12
Natural Gas Prod.	Bromoform	secondary	-1.51e-10	-7.64e-12
Natural Gas Prod.	1,2-Dichloroethane	secondary	-1.54e-10	-7.80e-12
Natural Gas Prod.	Barium cmpds	secondary	-1.61e-10	-8.15e-12
CRT Incineration	Benzo[b,j,k]fluoranthene	secondary	-1.64e-10	-8.29e-12
Natural Gas Prod.	Methyl methacrylate	secondary	-1.85e-10	-9.36e-12
Natural Gas Prod.	Silicon	secondary	-1.86e-10	-9.43e-12
Fuel Oil #4 Prod.	Ethylbenzene	secondary	-1.99e-10	-1.01e-11
Natural Gas Prod.	Tetrachloroethylene	secondary	-2.13e-10	-1.08e-11
Natural Gas Prod.	Methyl ethyl ketone	secondary	-2.16e-10	-1.09e-11
Fuel Oil #4 Prod.	o-xylene	secondary	-2.16e-10	-1.10e-11
Natural Gas Prod.	Chlorine	secondary	-2.37e-10	-1.20e-11
CRT Incineration	Acenaphthene	secondary	-2.58e-10	-1.31e-11
Natural Gas Prod.	Zinc (+2)	secondary	-2.63e-10	-1.33e-11
Natural Gas Prod.	Isophorone	secondary	-2.68e-10	-1.36e-11
Natural Gas Prod.	Dimethyl sulfate	secondary	-2.79e-10	-1.42e-11
Fuel Oil #4 Prod.	Chromium (III)	secondary	-3.09e-10	-1.57e-11
Fuel Oil #4 Prod.	Copper (+1 & +2)	secondary	-3.21e-10	-1.63e-11
Fuel Oil #4 Prod.	Aromatic hydrocarbons	secondary	-3.43e-10	-1.74e-11
CRT Incineration	Acenaphthylene	secondary	-3.72e-10	-1.89e-11
Fuel Oil #4 Prod.	2,4-Dinitrotoluene	secondary	-4.01e-10	-2.03e-11
Fuel Oil #4 Prod.	Chlorobenzene	secondary	-5.04e-10	-2.55e-11
Natural Gas Prod.	Di(2-ethylhexyl)phthalate	secondary	-5.84e-10	-2.96e-11
CRT Incineration	Biphenyl	secondary	-6.02e-10	-3.05e-11
Fuel Oil #4 Prod.	Dichloromethane	secondary	-6.02e-10	-3.05e-11
Fuel Oil #4 Prod.	Bromoform	secondary	-6.24e-10	-3.16e-11
CRT Incineration	Acetophenone	secondary	-6.27e-10	-3.18e-11
Fuel Oil #4 Prod.	1,2-Dichloroethane	secondary	-6.36e-10	-3.22e-11
Natural Gas Prod.	Cadmium	secondary	-7.04e-10	-3.56e-11
Natural Gas Prod.	Propionaldehyde	secondary	-7.60e-10	-3.85e-11
Natural Gas Prod.	Acetaldehyde	secondary	-7.60e-10	-3.85e-11
Fuel Oil #4 Prod.	Methyl methacrylate	secondary	-7.64e-10	-3.87e-11
Fuel Oil #4 Prod.	Tetrachloroethylene	secondary	-8.79e-10	-4.46e-11
Fuel Oil #4 Prod.	Methyl ethyl ketone	secondary	-8.93e-10	-4.53e-11
Natural Gas Prod.	Mercury	secondary	-9.44e-10	-4.78e-11
Natural Gas Prod.	Methyl hydrazine	secondary	-9.89e-10	-5.01e-11
CRT Incineration	Cumene	secondary	-1.01e-09	-5.11e-11
Fuel Oil #4 Prod.	1,4-Dichlorobenzene	secondary	-1.08e-09	-5.47e-11
CRT Incineration	Aluminum (elemental)	secondary	-1.08e-09	-5.49e-11
Fuel Oil #4 Prod.	Isophorone	secondary	-1.11e-09	-5.61e-11

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Fuel Oil #4 Prod.	Dimethyl sulfate	secondary	-1.15e-09	-5.85e-11
CRT Incineration	Ethyl Chloride	secondary	-1.19e-09	-6.04e-11
Natural Gas Prod.	Cobalt	secondary	-1.32e-09	-6.71e-11
Natural Gas Prod.	Methyl chloride	secondary	-1.37e-09	-6.93e-11
Natural Gas Prod.	Beryllium	secondary	-1.40e-09	-7.11e-11
Natural Gas Prod.	Aluminum (+3)	secondary	-1.44e-09	-7.27e-11
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	-1.61e-09	-8.18e-11
CRT Incineration	1,1,1-Trichloroethane	secondary	-1.68e-09	-8.52e-11
Natural Gas Prod.	Acrolein	secondary	-1.69e-09	-8.55e-11
Fuel Oil #4 Prod.	Cadmium cmpds	secondary	-1.70e-09	-8.62e-11
CRT Incineration	Ethylene dibromide	secondary	-1.78e-09	-9.04e-11
Fuel Oil #4 Prod.	Toluene	secondary	-2.04e-09	-1.03e-10
CRT Incineration	Xylene (mixed isomers)	secondary	-2.13e-09	-1.08e-10
Natural Gas Prod.	o-xylene	secondary	-2.31e-09	-1.17e-10
Fuel Oil #4 Prod.	Di(2-ethylhexyl)phthalate	secondary	-2.41e-09	-1.22e-10
Natural Gas Prod.	PM-10	secondary	-2.48e-09	-1.25e-10
Fuel Oil #4 Prod.	Cadmium	secondary	-2.87e-09	-1.45e-10
Fuel Oil #4 Prod.	Mercury	secondary	-3.11e-09	-1.57e-10
Fuel Oil #4 Prod.	Propionaldehyde	secondary	-3.14e-09	-1.59e-10
Fuel Oil #4 Prod.	Acetaldehyde	secondary	-3.14e-09	-1.59e-10
Natural Gas Prod.	Chloroform	secondary	-3.17e-09	-1.60e-10
CRT Incineration	Phenanthrene	secondary	-4.02e-09	-2.04e-10
Natural Gas Prod.	Benzyl chloride	secondary	-4.07e-09	-2.06e-10
Fuel Oil #4 Prod.	Methyl hydrazine	secondary	-4.09e-09	-2.07e-10
Natural Gas Prod.	Toluene	secondary	-4.33e-09	-2.19e-10
CRT Incineration	Vinyl acetate	secondary	-4.41e-09	-2.23e-10
CRT Incineration	Styrene	secondary	-4.52e-09	-2.29e-10
CRT Incineration	Silicon	secondary	-4.73e-09	-2.40e-10
Fuel Oil #4 Prod.	Beryllium	secondary	-5.02e-09	-2.54e-10
Natural Gas Prod.	Carbon disulfide	secondary	-5.20e-09	-2.63e-10
Fuel Oil #4 Prod.	Methyl chloride	secondary	-5.65e-09	-2.86e-10
CRT Incineration	Methyl tert-butyl ether	secondary	-6.19e-09	-3.14e-10
Fuel Oil #4 Prod.	Acrolein	secondary	-6.98e-09	-3.53e-10
Natural Gas Prod.	Phenol	secondary	-7.32e-09	-3.71e-10
Natural Gas Prod.	Nitrate	secondary	-7.54e-09	-3.82e-10
Fuel Oil #4 Prod.	Cobalt	secondary	-9.27e-09	-4.69e-10
CRT Incineration	Chloroacetophenone	secondary	-1.04e-08	-5.27e-10
Natural Gas Prod.	Lead	secondary	-1.09e-08	-5.50e-10
CRT Incineration	Ethylbenzene	secondary	-1.18e-08	-5.97e-10
Fuel Oil #4 Prod.	Chloroform	secondary	-1.31e-08	-6.63e-10
Natural Gas Prod.	Cyanide (-1)	secondary	-1.45e-08	-7.37e-10
Fuel Oil #4 Prod.	Aluminum (elemental)	secondary	-1.61e-08	-8.15e-10
Fuel Oil #4 Prod.	Benzyl chloride	secondary	-1.68e-08	-8.53e-10
CRT Incineration	Chromium (III)	secondary	-1.69e-08	-8.58e-10
Natural Gas Prod.	Manganese	secondary	-1.98e-08	-1.00e-09
Fuel Oil #4 Prod.	Carbon disulfide	secondary	-2.15e-08	-1.09e-09
CRT Incineration	Aluminum (+3)	secondary	-2.23e-08	-1.13e-09
CRT Incineration	2,4-Dinitrotoluene	secondary	-2.48e-08	-1.25e-09
Natural Gas Prod.	Copper	secondary	-2.65e-08	-1.34e-09
Natural Gas Prod.	Bromomethane	secondary	-2.77e-08	-1.40e-09
		-		

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Natural Gas Prod.	2-Chloroacetophenone	secondary	-2.80e-08	-1.42e-09
Natural Gas Prod.	Barium	secondary	-2.96e-08	-1.50e-09
CRT Incineration	Chlorobenzene	secondary	-3.11e-08	-1.58e-09
Natural Gas Prod.	Hydrogen sulfide	secondary	-3.37e-08	-1.71e-09
CRT Incineration	Dichloromethane	secondary	-3.70e-08	-1.87e-09
CRT Incineration	1,2-Dichloroethane	secondary	-3.83e-08	-1.94e-09
CRT Incineration	Bromoform	secondary	-3.85e-08	-1.95e-09
Fuel Oil #4 Prod.	Barium cmpds	secondary	-3.89e-08	-1.97e-09
Fuel Oil #4 Prod.	Lead	secondary	-4.45e-08	-2.25e-09
CRT Incineration	Methyl methacrylate	secondary	-4.72e-08	-2.39e-09
Natural Gas Prod.	Chromium (VI)	secondary	-4.74e-08	-2.40e-09
CRT Incineration	Dimethylbenzanthracene	secondary	-4.82e-08	-2.44e-09
CRT Incineration	Tetrachloroethylene	secondary	-5.31e-08	-2.69e-09
Fuel Oil #4 Prod.	Barium	secondary	-5.46e-08	-2.77e-09
CRT Incineration	Methyl ethyl ketone	secondary	-5.52e-08	-2.80e-09
Fuel Oil #4 Prod.	Cyanide (-1)	secondary	-6.02e-08	-3.05e-09
CRT Incineration	Cadmium	secondary	-6.06e-08	-3.07e-09
Natural Gas Prod.	Nickel	secondary	-6.16e-08	-3.12e-09
CRT Incineration	PM-10	secondary	-6.30e-08	-3.19e-09
Natural Gas Prod.	Naphthalene	secondary	-6.53e-08	-3.31e-09
CRT Incineration	Toluene	secondary	-6.65e-08	-3.37e-09
CRT Incineration	Isophorone	secondary	-6.84e-08	-3.47e-09
Fuel Oil #4 Prod.	Silicon	secondary	-7.03e-08	-3.56e-09
CRT Incineration	Dimethyl sulfate	secondary	-7.14e-08	-3.62e-09
Fuel Oil #4 Prod.	Manganese	secondary	-8.00e-08	-4.05e-09
Fuel Oil #4 Prod.	Naphthalene	secondary	-8.01e-08	-4.06e-09
Fuel Oil #4 Prod.	Dimethylbenzanthracene	secondary	-9.64e-08	-4.88e-09
CRT Incineration	Mercury	secondary	-1.12e-07	-5.66e-09
Fuel Oil #4 Prod.	Bromomethane	secondary	-1.15e-07	-5.80e-09
Fuel Oil #4 Prod.	2-Chloroacetophenone	secondary	-1.16e-07	-5.86e-09
CRT Incineration	Phenol	secondary	-1.17e-07	-5.94e-09
Fuel Oil #4 Prod.	Copper	secondary	-1.34e-07	-6.79e-09
Natural Gas Prod.	Dimethylbenzanthracene	secondary	-1.44e-07	-7.30e-09
CRT Incineration	Di(2-ethylhexyl)phthalate	secondary	-1.49e-07	-7.55e-09
Fuel Oil #4 Prod.	Nitrate	secondary	-1.58e-07	-7.99e-09
Fuel Oil #4 Prod.	Chromium (VI)	secondary	-1.82e-07	-9.21e-09
CRT Incineration	Cobalt	secondary	-1.92e-07	-9.71e-09
CRT Incineration	Acetaldehyde	secondary	-1.94e-07	-9.83e-09
CRT Incineration	Propionaldehyde	secondary	-1.94e-07	-9.83e-09
Natural Gas Prod.	Antimony	secondary	-2.23e-07	-1.13e-08
CRT Incineration	Methyl hydrazine	secondary	-2.53e-07	-1.28e-08
CRT Incineration	Beryllium	secondary	-3.00e-07	-1.52e-08
Fuel Oil #4 Prod.	Aluminum (+3)	secondary	-3.47e-07	-1.76e-08
CRT Incineration	Methyl chloride	secondary	-3.49e-07	-1.77e-08
CRT Incineration	Copper	secondary	-4.12e-07	-2.09e-08
CRT Incineration	Acrolein	secondary	-4.31e-07	-2.18e-08
CRT Incineration	Hydrogen sulfide	secondary	-6.11e-07	-3.10e-08
Fuel Oil #4 Prod.	Antimony	secondary	-6.37e-07	-3.23e-08
CRT Incineration	Hexane	secondary	-7.53e-07	-3.81e-08
CRT Incineration	Chloroform	secondary	-7.95e-07	-4.03e-08

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Hydrofluoric acid	secondary	-8.74e-07	-4.43e-08
Natural Gas Prod.	Phosphorus (yellow or white)	secondary	-8.87e-07	-4.49e-08
Fuel Oil #4 Prod.	PM-10	secondary	-9.35e-07	-4.74e-08
CRT Incineration	Pentane	secondary	-9.44e-07	-4.78e-08
CRT Incineration	Benzyl chloride	secondary	-1.04e-06	-5.27e-08
CRT Incineration	Ethane	secondary	-1.13e-06	-5.70e-08
Fuel Oil #4 Prod.	Nickel	secondary	-1.16e-06	-5.87e-08
Natural Gas Prod.	Molybdenum	secondary	-1.16e-06	-5.87e-08
CRT Incineration	Barium	secondary	-1.31e-06	-6.66e-08
CRT Incineration	Carbon disulfide	secondary	-1.33e-06	-6.73e-08
Fuel Oil #4 Prod.	Hexane	secondary	-1.36e-06	-6.90e-08
Fuel Oil #4 Prod.	Phenol	secondary	-1.75e-06	-8.88e-08
CRT Incineration	Nitrate	secondary	-1.80e-06	-9.11e-08
Fuel Oil #4 Prod.	Pentane	secondary	-1.97e-06	-9.96e-08
CRT Incineration	Naphthalene	secondary	-2.03e-06	-1.03e-07
Natural Gas Prod.	Hexane	secondary	-2.03e-06	-1.03e-07
Fuel Oil #4 Prod.	Ethane	secondary	-2.34e-06	-1.19e-07
Natural Gas Prod.	Pentane	secondary	-2.94e-06	-1.49e-07
Natural Gas Prod.	Ethane	secondary	-3.50e-06	-1.77e-07
Fuel Oil #4 Prod.	Zinc (elemental)	secondary	-3.56e-06	-1.80e-07
Fuel Oil #4 Prod.	Hydrofluoric acid	secondary	-3.61e-06	-1.83e-07
CRT Incineration	Cyanide (-1)	secondary	-3.72e-06	-1.88e-07
Natural Gas Prod.	Zinc (elemental)	secondary	-4.26e-06	-2.16e-07
CRT Incineration	Manganese	secondary	-4.89e-06	-2.48e-07
Natural Gas Prod.	Nitrous oxide	secondary	-5.79e-06	-2.93e-07
Natural Gas Prod.	Selenium	secondary	-6.09e-06	-3.08e-07
Fuel Oil #4 Prod.	Molybdenum	secondary	-6.48e-06	-3.28e-07
Natural Gas Prod.	Fluorides (F-)	secondary	-6.78e-06	-3.44e-07
CRT Incineration	Bromomethane	secondary	-7.08e-06	-3.59e-07
CRT Incineration	Nickel	secondary	-9.71e-06	-4.92e-07
CRT Incineration	Chromium (VI)	secondary	-9.94e-06	-5.04e-07
CRT Incineration	Molybdenum	secondary	-9.95e-06	-5.04e-07
Fuel Oil #4 Prod.	Ammonia	secondary	-1.10e-05	-5.57e-07
Fuel Oil #4 Prod.	Hydrogen sulfide	secondary	-1.27e-05	-6.45e-07
Natural Gas Prod.	Formaldehyde	secondary	-1.39e-05	-7.04e-07
CRT Incineration	Antimony	secondary	-2.16e-05	-1.09e-06
CRT Incineration	Zinc (elemental)	secondary	-2.54e-05	-1.29e-06
Fuel Oil #4 Prod.	Fluorides (F-)	secondary	-2.64e-05	-1.34e-06
Fuel Oil #4 Prod.	Selenium	secondary	-2.69e-05	-1.36e-06
Natural Gas Prod.	Hydrochloric acid	secondary	-3.20e-05	-1.62e-06
Natural Gas Prod.	Vanadium	secondary	-3.49e-05	-1.77e-06
Fuel Oil #4 Prod.	Phosphorus (yellow or white)	secondary	-3.98e-05	-2.02e-06
CRT Incineration	Ammonia	secondary	-4.51e-05	-2.28e-06
CRT Incineration	Phosphorus (yellow or white)	secondary	-5.50e-05	-2.78e-06
Fuel Oil #4 Prod.	Nitrous oxide	secondary	-6.18e-05	-3.13e-06
Fuel Oil #4 Prod.	Hydrochloric acid	secondary	-1.32e-04	-6.70e-06
Natural Gas Prod.	Ammonia	secondary	-1.35e-04	-6.83e-06
Natural Gas Prod.	Sulfur oxides	secondary	-1.38e-04	-7.00e-06
Natural Gas Prod.	Arsenic	secondary	-1.58e-04	-7.99e-06
Natural Gas Prod.	PM	secondary	-2.08e-04	-1.05e-05

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
CRT Incineration	Hydrofluoric acid	secondary	-2.23e-04	-1.13e-05
Fuel Oil #4 Prod.	PM	secondary	-4.35e-04	-2.20e-05
CRT Incineration	Formaldehyde	secondary	-4.47e-04	-2.26e-05
Fuel Oil #4 Prod.	Formaldehyde	secondary	-5.52e-04	-2.80e-05
Fuel Oil #4 Prod.	Arsenic	secondary	-7.57e-04	-3.83e-05
CRT Incineration	Nitrous oxide	secondary	-1.04e-03	-5.27e-05
CRT Incineration	Vanadium	secondary	-1.17e-03	-5.94e-05
CRT Incineration	Selenium	secondary	-1.53e-03	-7.76e-05
CRT Incineration	Fluorides (F-)	secondary	-1.56e-03	-7.91e-05
Fuel Oil #4 Prod.	Nitrogen oxides	secondary	-1.93e-03	-9.77e-05
Fuel Oil #4 Prod.	Benzene	secondary	-2.59e-03	-1.31e-04
Fuel Oil #4 Prod.	Sulfur oxides	secondary	-2.63e-03	-1.33e-04
Fuel Oil #4 Prod.	Methane	secondary	-2.80e-03	-1.42e-04
CRT Incineration	Benzene	secondary	-3.18e-03	-1.61e-04
Fuel Oil #4 Prod.	Vanadium	secondary	-4.33e-03	-2.19e-04
CRT Incineration	Hydrochloric acid	secondary	-4.66e-03	-2.36e-04
Natural Gas Prod.	Nitrogen oxides	secondary	-4.68e-03	-2.37e-04
CRT Incineration	Nitrogen oxides	secondary	-1.25e-02	-6.36e-04
CRT Incineration	Carbon monoxide	secondary	-1.44e-02	-7.31e-04
Fuel Oil #4 Prod.	Carbon monoxide	secondary	-1.51e-02	-7.64e-04
CRT Incineration	Methane	secondary	-1.52e-02	-7.68e-04
CRT Incineration	PM	secondary	-1.83e-02	-9.29e-04
Natural Gas Prod.	Methane	secondary	-2.52e-02	-1.27e-03
CRT Incineration	Sulfur oxides	secondary	-2.70e-02	-1.37e-03
Natural Gas Prod.	Carbon monoxide	secondary	-3.19e-02	-1.62e-03
Natural Gas Prod.	Benzene	secondary	-3.79e-02	-1.92e-03
CRT Incineration	Arsenic	secondary	-3.84e-02	-1.95e-03
Total End-of-life			2.88e-01	1.46e-02
Total All Life-cycle Stages			1.97e+03	1.00e+02

 Table M-39. CRT LCIA Results for the Terrestrial Toxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Materials Processing Life-cycle St	tage			
Steel Prod., cold-rolled, semi-finished	Sulfur dioxide	secondary	1.24e+01	1.39e+00
PMMA Sheet Prod.	Sulfur dioxide	secondary	8.61e+00	9.63e-01
Natural Gas Prod.	Benzene	secondary	5.28e+00	5.91e-01
Aluminum Prod.	Sulfur dioxide	secondary	5.13e+00	5.74e-01
Natural Gas Prod.	Carbon monoxide	secondary	4.44e+00	4.97e-01
Polycarbonate Production	Sulfur dioxide	secondary	4.43e+00	4.95e-01
Natural Gas Prod.	Methane	secondary	3.50e+00	3.92e-01
Natural Gas Prod.	Nitrogen oxides	secondary	6.51e-01	7.28e-02
Aluminum Prod.	Titanium tetrachloride	secondary	2.68e-01	3.00e-02
Steel Prod., cold-rolled, semi-finished	Carbon monoxide	secondary	1.36e-01	1.52e-02
Aluminum Prod.	Manganese cmpds	secondary	6.03e-02	6.74e-03
Steel Prod., cold-rolled, semi-finished	PM	secondary	5.14e-02	5.75e-03
Aluminum Prod.	Vanadium	secondary	4.00e-02	4.47e-03
Natural Gas Prod.	PM	secondary	2.89e-02	3.23e-03
PET Resin Production	Carbon monoxide	secondary	2.61e-02	2.92e-03
Steel Prod., cold-rolled, semi-finished	Vanadium	secondary	2.53e-02	2.82e-03
PMMA Sheet Prod.	Carbon monoxide	secondary	2.35e-02	2.62e-03
Polycarbonate Production	Carbon monoxide	secondary	2.32e-02	2.59e-03
Natural Gas Prod.	Arsenic	secondary	2.20e-02	2.46e-03
Natural Gas Prod.	Sulfur oxides	secondary	1.93e-02	2.15e-03
Natural Gas Prod.	Ammonia	secondary	1.88e-02	2.10e-03
Aluminum Prod.	Arsenic cmpds	secondary	1.77e-02	1.98e-03
PMMA Sheet Prod.	Methane	secondary	1.15e-02	1.29e-03
Polycarbonate Production	Methane	secondary	1.13e-02	1.27e-03
Polycarbonate Production	Nitrogen dioxide	secondary	1.08e-02	1.21e-03
PMMA Sheet Prod.	Nitrogen dioxide	secondary	1.07e-02	1.20e-03
Steel Prod., cold-rolled, semi-finished	Fluorides (F-)	secondary	9.36e-03	1.05e-03
Steel Prod., cold-rolled, semi-finished	Manganese cmpds	secondary	7.30e-03	8.17e-04
Styrene-butadiene Copolymer Prod.	Carbon monoxide	secondary	7.00e-03	7.83e-04
Steel Prod., cold-rolled, semi-finished	Nitrogen dioxide	secondary	6.86e-03	7.67e-04
Aluminum Prod.	Carbon monoxide	secondary	6.42e-03	7.18e-04
Steel Prod., cold-rolled, semi-finished	Methane	secondary	5.34e-03	5.97e-04
Natural Gas Prod.	Vanadium	secondary	4.86e-03	5.44e-04
Natural Gas Prod.	Hydrochloric acid	secondary	4.46e-03	4.99e-04
PET Resin Production	Sulfur oxides	secondary	3.99e-03	4.47e-04
Aluminum Prod.	Barium cmpds	secondary	3.89e-03	4.35e-04
Polycarbonate Production	PM	secondary	3.61e-03	4.04e-04
Aluminum Prod.	PM	secondary	3.31e-03	3.71e-04
PMMA Sheet Prod.	PM	secondary	3.22e-03	3.60e-04
Styrene-butadiene Copolymer Prod.	Nitrogen oxides	secondary	3.06e-03	3.42e-04
Styrene-butadiene Copolymer Prod.	Methane	secondary	3.00e-03	3.36e-04
Aluminum Prod.	Methane	secondary	3.00e-03	3.35e-04
Aluminum Prod.	Nitrogen dioxide	secondary	2.66e-03	2.97e-04
Styrene-butadiene Copolymer Prod.	Sulfur oxides	secondary	2.50e-03	2.79e-04
Aluminum Prod.	Selenium	secondary	2.34e-03	2.62e-04
Steel Prod., cold-rolled, semi-finished	Arsenic	secondary	2.22e-03	2.48e-04
Steel Prod., cold-rolled, semi-finished	Phosphorus (yellow or white)	secondary	1.99e-03	2.23e-04
Natural Gas Prod.	Formaldehyde	secondary	1.93e-03	2.16e-04
PET Resin Production	Nitrogen oxides	secondary	1.72e-03	1.93e-04

 Table M-40.
 LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
PMMA Sheet Prod.	Ammonia	secondary	1.04e-03	1.16e-04
Steel Prod., cold-rolled, semi-finished	Hydrochloric acid	secondary	1.00e-03	1.12e-04
PET Resin Production	Methane	secondary	9.98e-04	1.12e-04
Natural Gas Prod.	Fluorides (F-)	secondary	9.45e-04	1.06e-04
Steel Prod., cold-rolled, semi-finished	Benzene	secondary	8.62e-04	9.64e-05
Steel Prod., cold-rolled, semi-finished	Barium cmpds	secondary	8.49e-04	9.49e-05
Natural Gas Prod.	Selenium	secondary	8.48e-04	9.48e-05
Natural Gas Prod.	Nitrous oxide	secondary	8.07e-04	9.02e-05
Steel Prod., cold-rolled, semi-finished	Arsenic cmpds	secondary	7.53e-04	8.42e-05
PET Resin Production	PM	secondary	6.99e-04	7.82e-05
Aluminum Prod.	Hydrochloric acid	secondary	6.64e-04	7.43e-05
Natural Gas Prod.	Zinc (elemental)	secondary	5.94e-04	6.64e-05
Aluminum Prod.	Aluminum (+3)	secondary	5.91e-04	6.61e-05
PMMA Sheet Prod.	Nitrates/nitrites	secondary	5.70e-04	6.38e-05
Natural Gas Prod.	Ethane	secondary	4.88e-04	5.46e-05
Aluminum Prod.	Cadmium cmpds	secondary	4.36e-04	4.88e-05
Styrene-butadiene Copolymer Prod.	PM	secondary	4.34e-04	4.85e-05
Natural Gas Prod.	Pentane	secondary	4.09e-04	4.58e-05
Steel Prod., cold-rolled, semi-finished	Titanium tetrachloride	secondary	4.08e-04	4.56e-05
Steel Prod., cold-rolled, semi-finished	Ammonia	secondary	3.91e-04	4.38e-05
Steel Prod., cold-rolled, semi-finished	Nitrous oxide	secondary	3.81e-04	4.26e-05
PMMA Sheet Prod.	Hydrogen cyanide	secondary	3.73e-04	4.17e-05
PMMA Sheet Prod.	Hydrochloric acid	secondary	2.99e-04	3.34e-05
Natural Gas Prod.	Hexane	secondary	2.83e-04	3.17e-05
Polycarbonate Production	Hydrochloric acid	secondary	2.60e-04	2.91e-05
Steel Prod., cold-rolled, semi-finished	Silicon	secondary	2.48e-04	2.77e-05
Aluminum Prod.	Benzene	secondary	2.12e-04	2.37e-05
Steel Prod., cold-rolled, semi-finished	Formaldehyde	secondary	2.03e-04	2.27e-05
Polycarbonate Production	Sulfuric acid	secondary	1.77e-04	1.98e-05
Aluminum Prod.	Barium sulfate	secondary	1.69e-04	1.89e-05
Natural Gas Prod.	Molybdenum	secondary	1.61e-04	1.81e-05
Steel Prod., cold-rolled, semi-finished	Titanium	secondary	1.60e-04	1.79e-05
Steel Prod., cold-rolled, semi-finished	Selenium	secondary	1.47e-04	1.64e-05
PET Resin Production	Hydrochloric acid	secondary	1.45e-04	1.63e-05
Polycarbonate Production	Mercury compounds	secondary	1.36e-04	1.52e-05
PMMA Sheet Prod.	Sulfuric acid	secondary	1.32e-04	1.47e-05
Natural Gas Prod.	Phosphorus (yellow or white)	secondary	1.24e-04	1.38e-05
Natural Gas Prod.	Hydrofluoric acid	secondary	1.22e-04	1.36e-05
Steel Prod., cold-rolled, semi-finished	Ethane	secondary	1.20e-04	1.35e-05
Steel Prod., cold-rolled, semi-finished	Zinc (elemental)	secondary	1.06e-04	1.19e-05
PMMA Sheet Prod.	Mercury compounds	secondary	1.01e-04	1.13e-05
Steel Prod., cold-rolled, semi-finished	Molybdenum	secondary	9.20e-05	1.03e-05
Aluminum Prod.	Hydrofluoric acid	secondary	8.95e-05	1.00e-05
Aluminum Prod.	Zinc (elemental)	secondary	8.55e-05	9.56e-06
Aluminum Prod.	Titanium	secondary	8.47e-05	9.48e-06
Aluminum Prod.	Copper (+1 & +2)	secondary	6.70e-05	7.50e-06
PET Resin Production	Sulfuric acid	secondary	6.23e-05	6.97e-06
Styrene-butadiene Copolymer Prod.	Sulfuric acid	secondary	6.21e-05	6.95e-06
Steel Prod., cold-rolled, semi-finished	Barium	secondary	6.03e-05	6.74e-06

 Table M-40. LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category
Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Aluminum Prod.	Nitrous oxide	secondary	5.14e-05	5.75e-06
Polycarbonate Production	Fluorides (F-)	secondary	5.11e-05	5.72e-06
Steel Prod., cold-rolled, semi-finished	Tin (Sn++, Sn4+)	secondary	5.11e-05	5.71e-06
Aluminum Prod.	Perfluoromethane	secondary	4.83e-05	5.40e-06
Styrene-butadiene Copolymer Prod.	Mercury compounds	secondary	4.76e-05	5.32e-06
Steel Prod., cold-rolled, semi-finished	Antimony	secondary	4.66e-05	5.21e-06
Polycarbonate Production	Aromatic hydrocarbons	secondary	4.59e-05	5.13e-06
Aluminum Prod.	Silicon	secondary	4.07e-05	4.55e-06
PMMA Sheet Prod.	Fluorine	secondary	3.80e-05	4.25e-06
Natural Gas Prod.	Antimony	secondary	3.10e-05	3.47e-06
Steel Prod., cold-rolled, semi-finished	Silver compounds	secondary	3.03e-05	3.38e-06
Styrene-butadiene Copolymer Prod.	Hydrochloric acid	secondary	2.98e-05	3.33e-06
Styrene-butadiene Copolymer Prod.	Aromatic hydrocarbons	secondary	2.97e-05	3.32e-06
Steel Prod., cold-rolled, semi-finished	Nitrates/nitrites	secondary	2.45e-05	2.74e-06
Steel Prod., cold-rolled, semi-finished	Boron	secondary	2.41e-05	2.70e-06
Steel Prod., cold-rolled, semi-finished	Mercury compounds	secondary	2.22e-05	2.48e-06
Natural Gas Prod.	Dimethylbenzanthracene	secondary	2.01e-05	2.24e-06
PET Resin Production	Fluorine	secondary	1.80e-05	2.01e-06
Styrene-butadiene Copolymer Prod.	Fluorides (F-)	secondary	1.79e-05	2.01e-06
Aluminum Prod.	Nitrate	secondary	1.70e-05	1.90e-06
Steel Prod., cold-rolled, semi-finished	Barium sulfate	secondary	1.68e-05	1.88e-06
Aluminum Prod.	Barium	secondary	1.46e-05	1.63e-06
Steel Prod., cold-rolled, semi-finished	Cadmium cmpds	secondary	1.32e-05	1.48e-06
Steel Prod., cold-rolled, semi-finished	Aluminum (elemental)	secondary	1.21e-05	1.35e-06
Steel Prod., cold-rolled, semi-finished	Acetylene	secondary	1.14e-05	1.28e-06
Steel Prod., cold-rolled, semi-finished	Hydrofluoric acid	secondary	9.28e-06	1.04e-06
Natural Gas Prod.	Naphthalene	secondary	9.09e-06	1.02e-06
Aluminum Prod.	Copper	secondary	8.87e-06	9.92e-07
Natural Gas Prod.	Nickel	secondary	8.58e-06	9.60e-07
Steel Prod., cold-rolled, semi-finished	Pentane	secondary	8.31e-06	9.29e-07
Steel Prod., cold-rolled, semi-finished	Aluminum (+3)	secondary	7.72e-06	8.64e-07
Aluminum Prod.	Nickel cmpds	secondary	7.09e-06	7.93e-07
Steel Prod., cold-rolled, semi-finished	Hydrogen sulfide	secondary	6.66e-06	7.45e-07
Aluminum Prod.	Nickel	secondary	6.62e-06	7.40e-07
Natural Gas Prod.	Chromium (VI)	secondary	6.61e-06	7.39e-07
Aluminum Prod.	Zinc (+2)	secondary	6.49e-06	7.25e-07
Steel Prod., cold-rolled, semi-finished	Copper	secondary	5.99e-06	6.70e-07
Polycarbonate Production	Copper $(+1 \& +2)$	secondary	5.79e-06	6.48e-07
Aluminum Prod.	Ammonia	secondary	5.76e-06	6.44e-07
Aluminum Prod.	Aromatic hydrocarbons	secondary	5.27e-06	5.89e-07
Polycarbonate Production	Phenol	secondary	5.22e-06	5.83e-07
Aluminum Prod.	Nitrites	secondary	5.03e-06	5.63e-07
Natural Gas Prod.	Hydrogen sulfide	secondary	4.70e-06	5.26e-07
PMMA Sheet Prod.	Copper $(+1 \& +2)$	secondary	4.31e-06	4.81e-07
Steel Prod., cold-rolled, semi-finished	Copper $(+1 \& +2)$	secondary	4.21e-06	4.71e-07
Natural Gas Prod.	Barium	secondary	4.12e-06	4.61e-07
Steel Prod., cold-rolled. semi-finished	Nickel	secondary	3.99e-06	4.46e-07
Aluminum Prod.	Aluminum (elemental)	secondary	3.92e-06	4.38e-07
Natural Gas Prod.	2-Chloroacetophenone	secondary	3.90e-06	4.36e-07
Natural Gas Prod.	Bromomethane	secondary	3.86e-06	4.32e-07
			00	

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
PMMA Sheet Prod.	Cyanide (-1)	secondary	3.83e-06	4.29e-07
Aluminum Prod.	Strontium (Sr II)	secondary	3.72e-06	4.16e-07
Natural Gas Prod.	Copper	secondary	3.69e-06	4.12e-07
Steel Prod., cold-rolled, semi-finished	Aromatic hydrocarbons	secondary	3.51e-06	3.93e-07
PMMA Sheet Prod.	Hydrofluoric acid	secondary	3.45e-06	3.86e-07
Aluminum Prod.	Fluoride	secondary	3.39e-06	3.79e-07
Aluminum Prod.	Lead cmpds	secondary	3.25e-06	3.63e-07
Aluminum Prod.	Vanadium (V3+, V5+)	secondary	3.03e-06	3.39e-07
Natural Gas Prod.	Manganese	secondary	2.76e-06	3.09e-07
Steel Prod., cold-rolled, semi-finished	Strontium (Sr II)	secondary	2.29e-06	2.56e-07
Steel Prod., cold-rolled, semi-finished	Hexane	secondary	2.17e-06	2.43e-07
Polycarbonate Production	Ammonia	secondary	2.07e-06	2.31e-07
Polycarbonate Production	Hydrofluoric acid	secondary	2.06e-06	2.31e-07
PET Resin Production	Copper (+1 & +2)	secondary	2.04e-06	2.28e-07
Styrene-butadiene Copolymer Prod.	Copper (+1 & +2)	secondary	2.03e-06	2.27e-07
Natural Gas Prod.	Cyanide (-1)	secondary	2.03e-06	2.27e-07
Aluminum Drod	Molybdenum (Mo II, Mo III, Mo IV, Mo V,	sacandam	1.840.06	2.062.07
PMMA Shoot Brod	Mo v1)	secondary	1.640-00	2.00e-07
PiviviA Sheet Prod.	Chaming (M)	secondary	1./3e-00	1.900-07
Steel Prod., cold-folled, semi-linished	Chromium (VI)	secondary	1.02e-00	1.81e-07
PMMA Sheet Prod.	Hydrogen sulfide	secondary	1.53e-06	1./2e-0/
Natural Gas Prod.		secondary	1.51e-06	1.69e-07
Aluminum Prod.	Hydrogen sulfide	secondary	1.40e-06	1.5/e-0/
Styrene-butadiene Copolymer Prod.	Nitrate	secondary	1.2/e-06	1.42e-07
Steel Prod., cold-rolled, semi-finished	Boric acid	secondary	1.06e-06	1.19e-07
Natural Gas Prod.	Nitrate	secondary	1.05e-06	1.18e-07
Polycarbonate Production	Halogenated hydrocarbons (unspecified)	secondary	1.03e-06	1.15e-07
Polycarbonate Production	Hydrogen sulfide	secondary	1.03e-06	1.15e-07
Natural Gas Prod.	Phenol	secondary	1.02e-06	1.14e-07
Steel Prod., cold-rolled, semi-finished	Bromine	secondary	9.94e-07	1.11e-07
PET Resin Production	Hydrofluoric acid	secondary	9.08e-07	1.01e-07
Steel Prod., cold-rolled, semi-finished	Uranium	secondary	8.25e-07	9.23e-08
PMMA Sheet Prod.	Aromatic hydrocarbons	secondary	7.67e-07	8.58e-08
Steel Prod., cold-rolled, semi-finished	Toluene	secondary	7.44e-07	8.32e-08
Natural Gas Prod.	Carbon disulfide	secondary	7.24e-07	8.10e-08
Aluminum Prod.	Toluene	secondary	6.75e-07	7.55e-08
Natural Gas Prod.	Toluene	secondary	6.03e-07	6.74e-08
Steel Prod., cold-rolled, semi-finished	Ethylene	secondary	6.02e-07	6.73e-08
Steel Prod., cold-rolled, semi-finished	Cyanide (-1)	secondary	5.92e-07	6.62e-08
Steel Prod., cold-rolled, semi-finished	Methanol	secondary	5.84e-07	6.54e-08
Natural Gas Prod.	Benzyl chloride	secondary	5.67e-07	6.35e-08
Steel Prod., cold-rolled, semi-finished	Xylene (mixed isomers)	secondary	5.50e-07	6.15e-08
Steel Prod., cold-rolled, semi-finished	Lead	secondary	5.40e-07	6.04e-08
Steel Prod., cold-rolled, semi-finished	Naphthalene	secondary	5.39e-07	6.02e-08
Polycarbonate Production	Chlorine	secondary	5.16e-07	5.77e-08
Aluminum Prod.	Xylene (mixed isomers)	secondary	5.13e-07	5.73e-08
Steel Prod., cold-rolled, semi-finished	Phenol	secondary	4.75e-07	5.31e-08
Steel Prod., cold-rolled, semi-finished	Heptane	secondary	4.57e-07	5.12e-08
PMMA Sheet Prod.	Nickel cmpds	secondary	4.56e-07	5.10e-08

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Aluminum Prod.	Lead	secondary	4.47e-07	5.00e-08
Natural Gas Prod.	Chloroform	secondary	4.41e-07	4.93e-08
Steel Prod., cold-rolled, semi-finished	Lead cmpds	secondary	4.33e-07	4.84e-08
PMMA Sheet Prod.	Chlorine	secondary	3.83e-07	4.29e-08
Styrene-butadiene Copolymer Prod.	Aluminum (+3)	secondary	3.62e-07	4.05e-08
Natural Gas Prod.	PM-10	secondary	3.45e-07	3.86e-08
Aluminum Prod.	Chromium (VI)	secondary	3.31e-07	3.71e-08
Natural Gas Prod.	o-xylene	secondary	3.21e-07	3.59e-08
Steel Prod., cold-rolled, semi-finished	Nickel cmpds	secondary	2.92e-07	3.26e-08
Steel Prod., cold-rolled, semi-finished	Zinc (+2)	secondary	2.60e-07	2.91e-08
Polycarbonate Production	Aluminum (+3)	secondary	2.58e-07	2.88e-08
Polycarbonate Production	Ethanethiol	secondary	2.58e-07	2.88e-08
Polycarbonate Production	Lead	secondary	2.58e-07	2.88e-08
Polycarbonate Production	Mercury	secondary	2.58e-07	2.88e-08
Polycarbonate Production	Nickel (+2)	secondary	2.58e-07	2.88e-08
Polycarbonate Production	Nitrate	secondary	2.58e-07	2.88e-08
Polycarbonate Production	Nitrous oxide	secondary	2.58e-07	2.88e-08
Polycarbonate Production	Zinc (+2)	secondary	2.58e-07	2.88e-08
Aluminum Prod.	Phenol	secondary	2.38e-07	2.66e-08
Natural Gas Prod.	Acrolein	secondary	2.35e-07	2.63e-08
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	2.25e-07	2.51e-08
Styrene-butadiene Copolymer Prod.	Phenol	secondary	2.15e-07	2.41e-08
Natural Gas Prod.	Aluminum (+3)	secondary	2.00e-07	2.24e-08
Steel Prod., cold-rolled, semi-finished	Cobalt	secondary	1.98e-07	2.22e-08
PET Resin Production	Phenol	secondary	1.98e-07	2.21e-08
Natural Gas Prod.	Beryllium	secondary	1.95e-07	2.19e-08
PMMA Sheet Prod.	Aluminum (+3)	secondary	1.92e-07	2.14e-08
PMMA Sheet Prod.	Ethanethiol	secondary	1.92e-07	2.14e-08
PMMA Sheet Prod.	Lead	secondary	1.92e-07	2.14e-08
PMMA Sheet Prod.	Mercury	secondary	1.92e-07	2.14e-08
PMMA Sheet Prod.	Nitrous oxide	secondary	1.92e-07	2.14e-08
PMMA Sheet Prod.	Zinc (+2)	secondary	1.92e-07	2.14e-08
Natural Gas Prod.	Methyl chloride	secondary	1.90e-07	2.13e-08
Natural Gas Prod.	Cobalt	secondary	1.85e-07	2.06e-08
PET Resin Production	Chlorine	secondary	1.82e-07	2.03e-08
PET Resin Production	Mercury	secondary	1.82e-07	2.03e-08
Styrene-butadiene Copolymer Prod.	Chlorine	secondary	1.81e-07	2.02e-08
Styrene-butadiene Copolymer Prod.	Hydrofluoric acid	secondary	1.81e-07	2.02e-08
Steel Prod., cold-rolled, semi-finished	Manganese	secondary	1.76e-07	1.97e-08
Aluminum Prod.	Acetic acid	secondary	1.57e-07	1.76e-08
PET Resin Production	Ammonia	secondary	1.56e-07	1.74e-08
Styrene-butadiene Copolymer Prod.	Ammonia	secondary	1.55e-07	1.74e-08
Aluminum Prod.	Cobalt	secondary	1.53e-07	1.71e-08
Steel Prod., cold-rolled, semi-finished	Vanadium (V3+, V5+)	secondary	1.45e-07	1.63e-08
Natural Gas Prod.	Methyl hydrazine	secondary	1.38e-07	1.54e-08
Steel Prod., cold-rolled, semi-finished	Boron (B III)	secondary	1.32e-07	1.48e-08
Natural Gas Prod.	Mercury	secondary	1.32e-07	1.47e-08
Aluminum Prod.	Cadmium	secondary	1.10e-07	1.23e-08
Natural Gas Prod.	Acetaldehvde	secondary	1.06e-07	1.18e-08
Natural Gas Prod.	Propionaldehvde	secondary	1.06e-07	1.18e-08
	1			

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Steel Prod., cold-rolled, semi-finished	Acetic acid	secondary	1.00e-07	1.12e-08
Natural Gas Prod.	Cadmium	secondary	9.80e-08	1.10e-08
Steel Prod., cold-rolled, semi-finished	Triethylene glycol	secondary	9.39e-08	1.05e-08
PET Resin Production	Aluminum (+3)	secondary	9.08e-08	1.01e-08
PET Resin Production	Hydrogen sulfide	secondary	9.08e-08	1.01e-08
PET Resin Production	Lead	secondary	9.08e-08	1.01e-08
PET Resin Production	Nickel (+2)	secondary	9.08e-08	1.01e-08
PET Resin Production	Nitrate	secondary	9.08e-08	1.01e-08
PET Resin Production	Nitrous oxide	secondary	9.08e-08	1.01e-08
PET Resin Production	Zinc (+2)	secondary	9.08e-08	1.01e-08
Styrene-butadiene Copolymer Prod.	Halogenated hydrocarbons (unspecified)	secondary	9.04e-08	1.01e-08
Styrene-butadiene Copolymer Prod.	Hydrogen sulfide	secondary	9.04e-08	1.01e-08
Styrene-butadiene Copolymer Prod.	Lead	secondary	9.04e-08	1.01e-08
Styrene-butadiene Copolymer Prod.	Mercury	secondary	9.04e-08	1.01e-08
Styrene-butadiene Copolymer Prod.	Nickel (+2)	secondary	9.04e-08	1.01e-08
Styrene-butadiene Copolymer Prod.	Nitrous oxide	secondary	9.04e-08	1.01e-08
Styrene-butadiene Copolymer Prod.	Zinc (+2)	secondary	9.04e-08	1.01e-08
Aluminum Prod.	Triethylene glycol	secondary	9.03e-08	1.01e-08
Steel Prod., cold-rolled, semi-finished	Morpholine	secondary	8.78e-08	9.82e-09
Steel Prod., cold-rolled, semi-finished	Strontium	secondary	8.69e-08	9.72e-09
Steel Prod., cold-rolled, semi-finished	Propylene	secondary	8.67e-08	9.70e-09
Steel Prod., cold-rolled, semi-finished	Cadmium	secondary	8.54e-08	9.55e-09
Natural Gas Prod.	Di(2-ethylhexyl)phthalate	secondary	8.13e-08	9.09e-09
Aluminum Prod.	Mercury	secondary	7.51e-08	8.40e-09
Steel Prod., cold-rolled, semi-finished	Acetaldehyde	secondary	7.50e-08	8.39e-09
Aluminum Prod.	Cvanide (-1)	secondary	7.05e-08	7.89e-09
Aluminum Prod.	HALON-1301	secondary	6.72e-08	7.51e-09
Steel Prod., cold-rolled, semi-finished	Rubidium ion (Rb+)	secondary	5.93e-08	6.64e-09
Steel Prod., cold-rolled, semi-finished	Fluorine	secondary	5.57e-08	6.23e-09
Aluminum Prod.	Chromium (III)	secondary	4.88e-08	5.46e-09
	Molvbdenum (Mo II, Mo III, Mo IV, Mo V,			
Steel Prod., cold-rolled, semi-finished	Mo VI)	secondary	4.44e-08	4.97e-09
Steel Prod., cold-rolled, semi-finished	Benzo[a]pyrene	secondary	4.37e-08	4.89e-09
Natural Gas Prod.	Dimethyl sulfate	secondary	3.89e-08	4.35e-09
Steel Prod., cold-rolled, semi-finished	Acetone	secondary	3.87e-08	4.33e-09
Natural Gas Prod.	Isophorone	secondary	3.73e-08	4.17e-09
Natural Gas Prod.	Zinc (+2)	secondary	3.66e-08	4.09e-09
Natural Gas Prod.	Chlorine	secondary	3.31e-08	3.70e-09
Natural Gas Prod.	Methyl ethyl ketone	secondary	3.01e-08	3.37e-09
Natural Gas Prod.	Tetrachloroethylene	secondary	2.96e-08	3.31e-09
Steel Prod., cold-rolled, semi-finished	Lanthanum	secondary	2.63e-08	2.95e-09
Natural Gas Prod.	Silicon	secondary	2.59e-08	2.90e-09
Natural Gas Prod.	Methyl methacrylate	secondary	2.57e-08	2.88e-09
Steel Prod., cold-rolled, semi-finished	Ethylbenzene	secondary	2.57e-08	2.88e-09
Steel Prod., cold-rolled, semi-finished	Beryllium	secondary	2.31e-08	2.58e-09
Natural Gas Prod.	Barium cmpds	secondary	2.24e-08	2.51e-09
Natural Gas Prod.	1,2-Dichloroethane	secondary	2.14e-08	2.40e-09
Natural Gas Prod.	Bromoform	secondary	2.10e-08	2.35e-09
Natural Gas Prod.	Dichloromethane	secondary	2.03e-08	2.27e-09

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Chlorobenzene	secondary	1.70e-08	1.90e-09
Aluminum Prod.	Strontium	secondary	1.67e-08	1.87e-09
Steel Prod., cold-rolled, semi-finished	Mercury	secondary	1.60e-08	1.79e-09
Steel Prod., cold-rolled, semi-finished	Thorium	secondary	1.41e-08	1.58e-09
Natural Gas Prod.	2,4-Dinitrotoluene	secondary	1.35e-08	1.51e-09
Steel Prod., cold-rolled, semi-finished	HALON-1301	secondary	1.23e-08	1.38e-09
Natural Gas Prod.	3-Methylcholanthrene	secondary	1.18e-08	1.32e-09
Natural Gas Prod.	Chromium (III)	secondary	1.13e-08	1.26e-09
Steel Prod., cold-rolled, semi-finished	Tin	secondary	8.21e-09	9.19e-10
Steel Prod., cold-rolled, semi-finished	Phosphorus pentoxide	secondary	7.88e-09	8.81e-10
Natural Gas Prod.	Ethylbenzene	secondary	6.75e-09	7.55e-10
Steel Prod., cold-rolled, semi-finished	Scandium	secondary	6.40e-09	7.16e-10
Natural Gas Prod.	Aluminum (elemental)	secondary	5.93e-09	6.64e-10
Steel Prod., cold-rolled, semi-finished	Chlorine	secondary	5.59e-09	6.25e-10
Natural Gas Prod.	Phenanthrene	secondary	4.92e-09	5.50e-10
Steel Prod., cold-rolled, semi-finished	Thallium	secondary	4.36e-09	4.87e-10
Aluminum Prod.	1,2-Dichlorotetrafluoroethane	secondary	3.88e-09	4.34e-10
Natural Gas Prod.	2-Methylnaphthalene	secondary	3.78e-09	4.23e-10
Natural Gas Prod.	Methyl tert-butyl ether	secondary	3.38e-09	3.78e-10
Aluminum Prod.	Perfluoroethane	secondary	3.15e-09	3.52e-10
Steel Prod., cold-rolled, semi-finished	Chromium (III)	secondary	2.59e-09	2.90e-10
Steel Prod., cold-rolled, semi-finished	Nitrites	secondary	2.54e-09	2.85e-10
Natural Gas Prod.	Styrene	secondary	2.46e-09	2.76e-10
Natural Gas Prod.	Vinyl acetate	secondary	2.40e-09	2.69e-10
Natural Gas Prod.	Cadmium cmpds	secondary	9.80e-10	1.10e-10
Natural Gas Prod.	Ethylene dibromide	secondary	9.73e-10	1.09e-10
Steel Prod., cold-rolled, semi-finished	Edetic acid (EDTA)	secondary	9.46e-10	1.06e-10
Natural Gas Prod.	1,1,1-Trichloroethane	secondary	9.17e-10	1.03e-10
Steel Prod., cold-rolled, semi-finished	Hypochlorous acid	secondary	8.62e-10	9.64e-11
Steel Prod., cold-rolled, semi-finished	Cobalt (Co I, Co II, Co III)	secondary	7.70e-10	8.61e-11
Steel Prod., cold-rolled, semi-finished	Dichloromethane	secondary	7.10e-10	7.94e-11
Natural Gas Prod.	Ethyl Chloride	secondary	6.50e-10	7.27e-11
Natural Gas Prod.	Cumene	secondary	5.50e-10	6.15e-11
Natural Gas Prod.	Acenaphthylene	secondary	4.89e-10	5.47e-11
Natural Gas Prod.	Benzo[a]anthracene	secondary	3.92e-10	4.38e-11
Natural Gas Prod.	Chrysene	secondary	3.77e-10	4.22e-11
Natural Gas Prod.	Indeno(1,2,3-cd)pyrene	secondary	3.44e-10	3.85e-11
Natural Gas Prod.	Acetophenone	secondary	3.42e-10	3.83e-11
Natural Gas Prod.	Biphenyl	secondary	3.28e-10	3.67e-11
Natural Gas Prod.	Acenaphthene	secondary	3.14e-10	3.52e-11
Natural Gas Prod.	Benzo[b]fluoranthene	secondary	2.91e-10	3.26e-11
Natural Gas Prod.	Benzo[g,h,i]perylene	secondary	2.23e-10	2.49e-11
Natural Gas Prod.	Benzo[a]pyrene	secondary	2.20e-10	2.46e-11
Natural Gas Prod.	Dibenzo[a,h]anthracene	secondary	1.98e-10	2.21e-11
Natural Gas Prod.	Aromatic hydrocarbons	secondary	1.98e-10	2.21e-11
Natural Gas Prod.	Copper (+1 & +2)	secondary	1.85e-10	2.07e-11
Natural Gas Prod.	Pyrene	secondary	1.71e-10	1.91e-11
Natural Gas Prod.	Fluorene	secondary	1.14e-10	1.28e-11
Natural Gas Prod.	Fluoranthene	secondary	1.02e-10	1.14e-11
Steel Prod., cold-rolled, semi-finished	Isopropylpropionate	secondary	9.91e-11	1.11e-11

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Benzo[b,j,k]fluoranthene	secondary	8.92e-11	9.97e-12
Steel Prod., cold-rolled, semi-finished	Lithium salts	secondary	4.86e-11	5.43e-12
Steel Prod., cold-rolled, semi-finished	Chloroform	secondary	2.65e-11	2.96e-12
Steel Prod., cold-rolled, semi-finished	Zirconium	secondary	2.28e-11	2.55e-12
Steel Prod., cold-rolled, semi-finished	Perfluoromethane	secondary	2.20e-11	2.46e-12
Natural Gas Prod.	5-Methyl chrysene	secondary	1.78e-11	1.99e-12
Natural Gas Prod.	Anthracene	secondary	6.68e-12	7.47e-13
Natural Gas Prod.	Nickel cmpds	secondary	1.96e-12	2.19e-13
Natural Gas Prod.	Lead cmpds	secondary	1.65e-12	1.84e-13
Natural Gas Prod.	Mercury compounds	secondary	9.98e-13	1.12e-13
Natural Gas Prod.	HALON-1301	secondary	7.17e-13	8.02e-14
Steel Prod., cold-rolled, semi-finished	Trichloroethylene	secondary	3.80e-13	4.25e-14
Natural Gas Prod.	Halogenated matter (organic)	secondary	1.65e-13	1.84e-14
Steel Prod., cold-rolled, semi-finished	Propionaldehyde	secondary	2.61e-14	2.91e-15
Steel Prod., cold-rolled, semi-finished	Tetrachloroethylene	secondary	1.07e-14	1.20e-15
Steel Prod., cold-rolled, semi-finished	Benzaldehyde	secondary	2.32e-15	2.59e-16
Steel Prod., cold-rolled, semi-finished	1,1,1-Trichloroethane	secondary	1.57e-15	1.76e-16
Natural Gas Prod.	Halogenated hydrocarbons (unspecified)	secondary	4.11e-16	4.60e-17
Steel Prod., cold-rolled, semi-finished	Hexachloroethane	secondary	6.02e-17	6.73e-18
Total Materials Processing			4.54e+01	5.08e+00
Manufacturing Life-cycle Stage				
Japanese Electric Grid	Sulfur dioxide	model/secondary	1.92e+02	2.14e+01
Monitor/module	Phosphine	primary	2.87e+01	3.21e+00
Monitor/module	Phosphorus (vellow or white)	primary	3.42e+00	3.82e-01
US electric grid	Sulfur dioxide	model/secondary	3.19e+00	3.57e-01
Monitor/module	Fluorides (F-)	primary	2.54e+00	2.84e-01
Monitor/module	Tetramethyl ammonium hydroxide	primary	6.43e-01	7.19e-02
Monitor/module	Nitrogen oxides	primary	5.48e-01	6.13e-02
Monitor/module	Hydrochloric acid	primary	2.77e-01	3.10e-02
LPG Production	Carbon monoxide	secondary	2.56e-01	2.87e-02
Monitor/module	Nitrogen fluoride	primary	2.45e-01	2.74e-02
Japanese Electric Grid	Nitrogen oxides	model/secondary	1.37e-01	1.53e-02
Natural Gas Prod.	Benzene	secondary	1.19e-01	1.33e-02
Japanese Electric Grid	Carbon monoxide	model/secondary	1.18e-01	1.32e-02
Monitor/module	Ammonia	primary	1.07e-01	1.20e-02
Natural Gas Prod.	Carbon monoxide	secondary	1.00e-01	1.12e-02
Natural Gas Prod.	Methane	secondary	7.91e-02	8.85e-03
Japanese Electric Grid	Vanadium	model/secondary	7.62e-02	8.53e-03
Monitor/module	Hydrofluoric acid	primary	5.21e-02	5.82e-03
LPG Production	Vanadium	secondary	5.02e-02	5.62e-03
LPG Production	Benzene	secondary	4.11e-02	4.60e-03
LPG Production	Methane	secondary	4.02e-02	4.50e-03
LPG Production	Sulfur oxides	secondary	3.85e-02	4.30e-03
Japanese Electric Grid	Arsenic	model/secondary	3.51e-02	3.92e-03
Backlight	Nitrogen oxides	primary	2.95e-02	3.29e-03
LPG Production	Nitrogen oxides	secondary	2.75e-02	3.07e-03
LCD glass mfg.	Fluorides (F-)	primary	2.70e-02	3.01e-03
Japanese Electric Grid	Hydrochloric acid	model/secondary	2.11e-02	2.36e-03
Panel components	Phosphorus (yellow or white)	primary	1.97e-02	2.20e-03

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Nitrogen oxides	secondary	1.47e-02	1.64e-03
LPG Production	Arsenic	secondary	9.91e-03	1.11e-03
Monitor/module	N-bromoacetamide	primary	9.18e-03	1.03e-03
Monitor/module	Sulfur hexafluoride	primary	7.30e-03	8.16e-04
Japanese Electric Grid	PM-10	model/secondary	6.72e-03	7.52e-04
LPG Production	Formaldehyde	secondary	6.43e-03	7.19e-04
LPG Production	PM	secondary	6.19e-03	6.92e-04
Monitor/module	Isopropyl alcohol	primary	4.55e-03	5.09e-04
Japanese Electric Grid	Fluorides (F-)	model/secondary	4.41e-03	4.93e-04
Japanese Electric Grid	Selenium	model/secondary	4.29e-03	4.80e-04
Japanese Electric Grid	Formaldehyde	model/secondary	3.93e-03	4.39e-04
US electric grid	Nitrogen oxides	model/secondary	2.28e-03	2.55e-04
Fuel Oil #4 Prod.	Carbon monoxide	secondary	2.16e-03	2.42e-04
Japanese Electric Grid	Zinc (elemental)	model/secondary	2.06e-03	2.31e-04
LCD glass mfg.	Nitrogen oxides	primary	2.04e-03	2.29e-04
PWB Mfg.	Formaldehyde	model/secondary	1.96e-03	2.19e-04
US electric grid	Carbon monoxide	model/secondary	1.96e-03	2.19e-04
LPG Production	Hydrochloric acid	secondary	1.77e-03	1.97e-04
Monitor/module	Arsenic	primary	1.70e-03	1.90e-04
Monitor/module	Monosilane	primary	1.54e-03	1.72e-04
US electric grid	Methane	model/secondary	1.25e-03	1.40e-04
Monitor/module	Sulfur oxides	primary	1.12e-03	1.25e-04
Japanese Electric Grid	Antimony	model/secondary	1.09e-03	1.22e-04
US electric grid	Arsenic	model/secondary	1.08e-03	1.21e-04
US electric grid	Hydrochloric acid	model/secondary	9.54e-04	1.07e-04
Fuel Oil #6 Prod.	Carbon monoxide	secondary	8.92e-04	9.98e-05
LPG Production	Nitrous oxide	secondary	7.78e-04	8.71e-05
Fuel Oil #2 Prod.	Carbon monoxide	secondary	7.27e-04	8.13e-05
Natural Gas Prod.	PM	secondary	6.53e-04	7.30e-05
Fuel Oil #4 Prod.	Vanadium	secondary	6.21e-04	6.94e-05
Monitor/module	Hexane	primary	5.88e-04	6.58e-05
Japanese Electric Grid	Hydrofluoric acid	model/secondary	5.77e-04	6.45e-05
Natural Gas Prod.	Arsenic	secondary	4.96e-04	5.55e-05
LPG Production	Phosphorus (yellow or white)	secondary	4.71e-04	5.27e-05
Japanese Electric Grid	Molybdenum	model/secondary	4.42e-04	4.94e-05
Natural Gas Prod.	Sulfur oxides	secondary	4.35e-04	4.86e-05
Natural Gas Prod.	Ammonia	secondary	4.24e-04	4.74e-05
Panel components	Nitrogen oxides	primary	4.11e-04	4.60e-05
Fuel Oil #4 Prod.	Methane	secondary	4.02e-04	4.49e-05
Fuel Oil #4 Prod.	Sulfur oxides	secondary	3.77e-04	4.22e-05
Japanese Electric Grid	Nitrous oxide	model/secondary	3.76e-04	4.20e-05
Fuel Oil #4 Prod.	Benzene	secondary	3.71e-04	4.15e-05
Fuel Oil #6 Prod.	Vanadium	secondary	3.67e-04	4.10e-05
LPG Production	Fluorides (F-)	secondary	3.60e-04	4.03e-05
LPG Production	Selenium	secondary	3.56e-04	3.98e-05
Monitor/module	Zinc (elemental)	primary	3.13e-04	3.50e-05
Japanese Electric Grid	Benzene	model/secondary	3.06e-04	3.42e-05
Fuel Oil #4 Prod.	Nitrogen oxides	secondary	2.77e-04	3.09e-05
Japanese Electric Grid	Methane	model/secondary	2.73e-04	3.06e-05
Monitor/module	Nitric acid	primary	2.68e-04	3.00e-05

 Table M-40. LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Fuel Oil # Prod. Methane secondary 2.01-04 2.24-05 Ded Oil # Prod Suffor oidse secondary 1.86-04 2.08-05 Monitorimodule Phosphoric acid primary 1.86-04 2.08-05 LG Production Amonoia secondary 1.78-04 2.01-045 Fuel Oil # Prod. Name secondary 1.61-04 1.88-045 Fuel Oil # Prod. Vandum secondary 1.46-04 1.88-045 Fuel Oil # Prod. Vandum secondary 1.40-04 1.56-05 Panel components HCFC-225ch primary 1.40-04 1.56-05 Fuel Oil # Prod. Methane secondary 1.38-04 1.31-05 Fuel Oil # Prod. Methane secondary 1.38-04 1.31-05 Fuel Oil # Prod. Methane secondary 1.80-04 1.32-05 Fuel Oil # Prod. Barenee secondary 1.80-04 1.32-05 Fuel Oil # Prod. Secondary 1.80-04 1.22-05 Natura Gae Prod.	Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Fuel Oli Ré Prod. Suffur oxides secondary 1.85.04 2.08-65 US electric grid Selenium model'secondary 1.80-04 2.01-65 LG Production Ammonia secondary 1.76-04 2.01-65 Eval Oli Ré Prod. Barcenco secondary 1.76-04 1.80-05 Fuel Oli Ré Prod. Nancains secondary 1.46-04 1.80-05 Fuel Oli Ré Prod. Nancains 1.67C-255cb primary 1.40-04 1.56-05 Fuel Oli Ré Prod. Nancain oxides secondary 1.38-04 1.51-045 Fuel Oli Ré Prod. Mandeum model'secondary 1.38-04 1.51-045 Fuel Oli Ré Prod. Mandeum secondary 1.18-04 1.32-045 Fuel Oli Ré Prod. Mandeum secondary 1.10-04 1.32-045 Fuel Oli Ré Prod. Mandeum secondary 1.10-04 1.32-045 Fuel Oli Ré Prod. Nareancia secondary 1.10-05 Natari Gas Prod. 1.32-045 Fuel Oli Ré Prod. Vanencia <	Fuel Oil #6 Prod.	Methane	secondary	2.01e-04	2.24e-05
Nontinomodule Pinosphoric acid primary 1.85-04 2.07-05 US electric grid Selenium model/accondary 1.78-04 2.010-05 Fuel OII #2 Prod. Benzene secondary 1.67-04 1.88-045 Fuel OII #2 Prod. Vanadium secondary 1.67-04 1.88-045 Fuel OII #2 Prod. Vanadium secondary 1.48-044 1.88-055 Panel components HCFC-225ch primary 1.40-044 1.56-055 Fuel OII #2 Prod. Nitrogen oxides secondary 1.38-044 1.51-045 Steletric grid Vanadium model/accondary 1.38-044 1.31-045 Fuel OII #2 Prod. Bazene secondary 1.18-044 1.32-045 Fuel OII #2 Prod. Battur oxides secondary 1.18-04 1.32-045 Staturi Gas Prod. Namatin secondary 1.18-04 1.32-045 Staturi Gas Prod. Americ secondary 1.18-04 1.32-045 Staturi Gas Prod. Americ secondary 1.18-04 <	Fuel Oil #6 Prod.	Sulfur oxides	secondary	1.86e-04	2.08e-05
US electric grid Selenium model'secondary 1.80-64 2.01-65 LPG Production Ammonia secondary 1.67-64 2.01-65 Pad Oil #6 Prod. Beazme secondary 1.67-64 1.86-645 Pad Oil #6 Prod. Wandum secondary 1.47-64 1.85-645 Panel components HCFC-225ca primary 1.40-64 1.55-645 Panel components HCFC-225cb primary 1.40-64 1.55-645 Panel Components HCFC-225cb primary 1.40-64 1.55-645 Panel Components HCFC-225cb primary 1.40-64 1.55-645 Panel Oil #2 Prod. Mandum model'secondary 1.38-64 1.35-645 Panel Oil #2 Prod. Mandum secondary 1.18-64 1.32-645 Panel Oil #2 Prod. Safur oxides secondary 1.18-64 1.32-645 Natural Gas Prod. Vandum secondary 1.18-64 1.22-645 Natural Gas Prod. Vanedia secondary 1.18-645 1.32-645	Monitor/module	Phosphoric acid	primary	1.85e-04	2.07e-05
LPG Production Ammonia secondary 1.79-e04 2.01-604 Fuel OH 26 Prod. Benzene secondary 1.61-e04 1.86-e05 LPG Prod. Vanafium secondary 1.61-e04 1.86-e05 Parel Components HCTC-22Sca primary 1.40-e04 1.56-e05 Parel components HCTC-22Sca primary 1.30-e04 1.56-e05 Fuel OH 76 Prod. Nitrogen oxides secondary 1.32-e04 1.56-e05 Fuel OH 72 Prod. Berzene secondary 1.32-e04 1.33-e05 Fuel OH 72 Prod. Berzene secondary 1.18-e04 1.32-e04 Fuel OH 72 Prod. Suffar oxides secondary 1.12-e04 1.32-e05 Fuel OH 72 Prod. Suffar oxides secondary 1.12-e04 1.32-e05 Fuel OH 72 Prod. Suffar oxides secondary 1.10-e04 1.32-e05 Suffar oxides secondary 1.10-e04 1.22-e05 Natural Gas Prod. Natural Gas Prod. 1.12-e04 1.12-e045 Natural Gas Prod. Natural	US electric grid	Selenium	model/secondary	1.80e-04	2.01e-05
Fuel Oil #6 Prod. Benzene secondary 1.67:004 1.86:005 Fuel Oil #2 Prod. Vanadium secondary 1.61:004 1.80:005 FUG Production Hydrogen sulfide secondary 1.40:004 1.56:005 Panel components HCTC-22Sca primary 1.40:004 1.56:05 Panel components HCTC-22Sca primary 1.30:004 1.56:05 Panel Oil #2 Prod. Methane secondary 1.20:004 1.33:005 Panel Oil #2 Prod. Methane secondary 1.19:004 1.33:005 Fuel Oil #2 Prod. Saffur oxides secondary 1.19:004 1.23:005 Fuel Oil #2 Prod. Saffur oxides secondary 1.10:044 1.23:005 Fuel Oil #2 Prod. Vanadium model/secondary 1.11:0:044 1.23:005 Natural Gas Prod. Hydrochloria acid primary 8.28:06 1.11:0:045 Natural Gas Prod. Hydrochloria acid primary 8.28:06 1.13:0:05 Natural Gas Prod. Norigorm oxides secondary	LPG Production	Ammonia	secondary	1.79e-04	2.01e-05
Fuel Oil 2 Prod. Vanatium secondary 1.61-04 1.30e-05 LPG Production Hydrogsn sulfide secondary 1.47e-04 1.65e-05 Panel components HCPC-225ca primary 1.40e-04 1.56e-05 Panel Components HCPC-225ch primary 1.40e-04 1.56e-05 Panel Oil 76 Prod. Nitrogen exoides secondary 1.32e-04 1.51e-05 Panel Oil 72 Prod. Betazene secondary 1.18e-04 1.33e-05 Panel Oil 72 Prod. Sulfar oxides secondary 1.18e-04 1.32e-05 Panel Oil 72 Prod. Sulfar oxides secondary 1.18e-04 1.22e-05 Panel Oil 74 Prod. Natural Gas Prod. Vanadium secondary 1.01e-04 1.23e-05 Natural Gas Prod. Arsenic secondary 1.01e-04 1.23e-05 Natural Gas Prod. Arsenic secondary 1.01e-04 1.23e-05 Natural Gas Prod. Arsenic secondary 1.01e-04 1.23e-05 Panel Oil 44 Prod. Arsenic <t< td=""><td>Fuel Oil #6 Prod.</td><td>Benzene</td><td>secondary</td><td>1.67e-04</td><td>1.86e-05</td></t<>	Fuel Oil #6 Prod.	Benzene	secondary	1.67e-04	1.86e-05
LPG Production Hydrogen sulfide secondary 1.47c-04 1.55c-05 Panel components HCFC-225ca primary 1.40c-04 1.56c-05 Panel components HCFC-225cb primary 1.39c-04 1.56c-05 Stectric grid Vandium model'scondary 1.39c-04 1.56c-05 Fuel Oil #2 Prod. Methane secondary 1.39c-04 1.34c-05 Fuel Oil #2 Prod. Benzene secondary 1.18c-04 1.33c-05 Japmese Electric Grid Nickel model'secondary 1.18c-04 1.23c-05 VS electric grid PM-10 model'secondary 1.18c-04 1.23c-05 Natural Gas Prod. Vanadium secondary 1.06c-04 1.21c-05 Japanese Electric Grid Bartum model'secondary 9.88c-05 1.10c-04 Japanese Electric Grid Bartum secondary 9.8c-05 9.25c-06 Fuel Oil #A Prod. Nortgen oxides secondary 7.9u-05 8.83c-06 Fuel Oil #A Prod. Formaldehyde secondary	Fuel Oil #2 Prod.	Vanadium	secondary	1.61e-04	1.80e-05
Panel components HCFC-225ch primary 1.40e-44 1.56e-05 Panel components HCFC-225ch primary 1.40e-44 1.56e-05 Evel Oil % Prod. Nitrogen oxides secondary 1.39e-04 1.51e-05 US lectric grid Vanadium model/secondary 1.39e-04 1.31e-05 Fuel Oil # Prod. Benzene secondary 1.19e-04 1.33e-05 Japanesz Electric Grid Nickel model/secondary 1.14e-04 1.23e-05 US electric grid PM-10 model/secondary 1.14e-04 1.23e-05 Natural Gas Prod. Arsenic secondary 1.08e-04 1.23e-05 Natural Gas Prod. Arsenic secondary 1.08e-04 1.28e-05 Japanesz Electric Grid Barium model/secondary 9.08e-05 1.10e-05 Japanesz Electric Grid Nitrogen oxides secondary 7.91e-05 8.84e-06 Fuel Oil # Prod. Arsenic secondary 7.91e-05 8.84e-06 Fuel Oil # Prod. Arsenic secondary	LPG Production	Hydrogen sulfide	secondary	1.47e-04	1.65e-05
Panel components ICC-225ch primary 1.40-04 1.56-05 Fuel Oil #6 Prod. Nitrogen oxides secondary 1.35-04 1.56-05 US electric grid Vandnium model/secondary 1.35-04 1.38-05 Fuel Oil #2 Prod. Berxone secondary 1.18-04 1.33-05 Japness Flectric Grid Nickel model/secondary 1.18-04 1.32-05 Stel Oil #2 Prod. Sulfar oxides secondary 1.18-04 1.23-05 Stel Oil #2 Prod. Sulfar oxides secondary 1.10-04 1.23-05 Stel Oil #2 Prod. Vandnium secondary 1.10-04 1.23-05 Natural Gas Prod. Hydrochloric acid secondary 1.00-04 1.23-05 Japanese Flectric Grid Barium model/secondary 9.86-05 1.10-04 Japanese Flectric Grid Barium secondary 9.86-05 1.10-05 Japanese Flectric Grid Nitrogen oxides secondary 9.86-05 1.10-05 Fuel Oil #2 Prod. Nitrogen oxides secondary </td <td>Panel components</td> <td>HCFC-225ca</td> <td>primary</td> <td>1.40e-04</td> <td>1.56e-05</td>	Panel components	HCFC-225ca	primary	1.40e-04	1.56e-05
Fuel Oil #6 Prod. Nirrogen oxides secondary 1.39c-04 1.51c-05 US electrie grid Vanadium model/secondary 1.35c-04 1.51c-05 Fuel Oil #2 Prod. Methane secondary 1.19c-04 1.33c-05 Japanese Electric Grid Nickel model/secondary 1.18c-04 1.32c-05 Japanese Electric Grid Nickel model/secondary 1.18c-04 1.22c-04 Janari Gas Prod. Vanadium secondary 1.18c-04 1.22c-05 Natural Gas Prod. Vanadium secondary 1.10c-04 1.23c-05 Japanese Electric Grid Arsenic secondary 1.00c-04 1.23c-05 Monitor/module Acetic acid primary 8.27c-05 9.25c-06 Fuel Oil #2 Prod. Nitrogen oxides secondary 7.90c-05 8.88c-06 LPG Production Molybdenum secondary 5.87c-05 6.47c-06 LPG Production Molybdenum secondary 5.87c-05 6.47c-06 LPG Production Zire (elemental) secondary	Panel components	HCFC-225cb	primary	1.40e-04	1.56e-05
US electric gridVanadiummodel/secondary1.35e-041.51e-05Fuel Olif /2 Prod.Methanesecondary1.10e-041.33e-05Japanese Electric GridNickelmodel/secondary1.18e-041.32e-05Japanese Electric GridNickelmodel/secondary1.18e-041.27e-05US electric gridPM-10model/secondary1.18e-041.27e-05Natural Gas Prod.Vanadiumsecondary1.18e-041.23e-05Japanese Electric GridArsenicsecondary1.08e-041.21e-05Natural Gas Prod.Hydrochloric acidsecondary9.88e-051.10e-04Japanese Electric GridBariummodel/secondary9.88e-059.17e-06Japanese Electric GridNotior/moduleAcetic acidprimary8.20e-059.17e-06Fuel Oli /4 Prod.Nitogen oxidessecondary7.91e-058.88e-06Fuel Oli /4 Prod.PMsecondary7.91e-058.84e-06Fuel Oli /4 Prod.PMsecondary5.75e-056.44e-06Japanese Electric GridAsptinhalenemodel/secondary5.75e-056.44e-06Japanese Electric GridNaphthalenemodel/secondary5.75e-056.44e-06Japanese Electric GridNaphthalenemodel/secondary5.75e-056.44e-06Japanese Electric GridNaphthalenemodel/secondary5.75e-056.44e-06Japanese Electric GridNaphthalenemodel/secondary3.75e-056.44e-06Japanese Electric	Fuel Oil #6 Prod.	Nitrogen oxides	secondary	1.39e-04	1.56e-05
Fuel Oil #2 Prod. Methane secondary 1.20e-04 1.33e-05 Fuel Oil #2 Prod. Benzene secondary 1.18e-04 1.33e-05 Japanese Electric Grid Nickel model'secondary 1.18e-04 1.32e-05 US cletric grid PM-10 model'secondary 1.12e-04 1.22e-05 Natural Gas Prod. Vanadium secondary 1.10e-04 1.22e-05 Natural Gas Prod. Arsenic secondary 1.08e-04 1.21e-05 Natural Gas Prod. Hydrochloric acid secondary 1.01e-04 1.13e-05 Japanese Electric Grid Bartinim model/secondary 9.8e-05 9.110e-05 Monitor/module Acetic acid primary 8.27e-05 9.25e-06 Fuel Oil # Prod. Formaldehyde secondary 7.9le-03 8.84e-06 LPG Production Molybdenum secondary 7.9le-03 8.84e-06 LPG Production Zinc (elemental) secondary 5.87e-05 6.44e-06 LPG Production Hydrofhoric acid secondary	US electric grid	Vanadium	model/secondary	1.35e-04	1.51e-05
Fuel Oil #2 Prod. Benzene secondary 1.19e-04 1.33e-05 Japanese Electric Grid Nickel model/secondary 1.18e-04 1.32e-05 US electric grid PM-10 model/secondary 1.14e-04 1.27e-05 Natural Gas Prod. Vanadium secondary 1.06e-04 1.23e-05 Natural Gas Prod. Hydrochloric acid secondary 1.06e-04 1.21e-05 Natural Gas Prod. Hydrochloric acid secondary 1.06e-04 1.21e-05 Japanese Electric Grid Barium model/secondary 9.88e-05 1.10e-04 1.13e-05 Japanese Electric Grid Barium model/secondary 8.20e-05 9.17e-06 Fuel Oil #2 Prod. Nitrogen oxides secondary 7.90e-05 8.84e-06 Fuel Oil #4 Prod. Formaldehyde secondary 7.90e-05 8.84e-06 Fuel Oil #4 Prod. Arsenic secondary 5.87e-05 6.44e-06 5.21e-06 6.44e-06 5.21e-06 Fuel Oil #6 Prod. Formaldehyde secondary 4.86e-05 5.21e-06 Natural Gas Prod. Formaldehyd	Fuel Oil #2 Prod.	Methane	secondary	1.20e-04	1.34e-05
Japanese Electric GridNickelmodel/secondary1.18e-041.32e-05Fuel Oli # 2 Prod.Sulfur oxidessecondary1.12e-041.23e-05Natural Gas Prod.Vanadiumsecondary1.10e-041.23e-05Jul # Prod.Arsenicsecondary1.01e-041.12e-05Japanese Electric GridBariummodel/secondary9.88e-051.10e-05Japanese Electric GridBariummodel/secondary9.88e-051.10e-05Monitor/moduleAcetic acidprimary8.27e-059.27e-06Fuel Oli # 2 Prod.Nitrogen oxidessecondary7.91e-058.88e-06Fuel Oli # 4 Prod.Formaldehydesecondary7.91e-058.88e-06Fuel Oli # 4 Prod.Formaldehydesecondary7.92e-058.84e-06Fuel Oli # 4 Prod.PMsecondary5.78e-056.57e-06Fuel Oli # 4 Prod.Arsenicsecondary5.78e-056.57e-06LPG ProductionMaphdalenemodel/secondary5.58e-056.44e-06Japanese Electric GridNaphthalenemodel/secondary4.88e-055.31e-06Fuel Oli # 6 Prod.Formaldehydesecondary4.86e-055.21e-06Natural Gas Prod.Formaldehydesecondary4.86e-055.21e-06Natural Gas Prod.Formaldehydesecondary3.32e-053.38e-05Fuel Oli # 6 Prod.Formaldehydesecondary3.32e-053.38e-05Fuel Oli # 6 Prod.Pormaldehydesecondary3.32e-053	Fuel Oil #2 Prod.	Benzene	secondary	1.19e-04	1.33e-05
Fuel Oil #2 Prod.Sulfur oxidessecondary1.14e-041.27e-05US electric gridPM-10model/secondary1.12e-041.25e-05Natural Gas Prod.Vanadiumsecondary1.08e-041.21e-05Natural Gas Prod.Arsenicsecondary1.08e-041.21e-05Japanese Electric GridBariummodel/secondary9.88e-051.10e-05Japanese Electric GridBariummodel/secondary8.27e-059.25e-06Fuel Oil #2 Prod.Nitrogen oxidessecondary7.91e-058.88e-06Fuel Oil #2 Prod.Formaldehydesecondary7.91e-058.88e-06Fuel Oil #4 Prod.Formaldehydesecondary7.90e-058.84e-06Fuel Oil #4 Prod.PMsecondary7.90e-058.84e-06Fuel Oil #4 Prod.PMsecondary5.87e-056.57e-06Fuel Oil #4 Prod.PMsecondary5.87e-056.15e-06LPG ProductionMolybdenumsecondary5.50e-056.15e-06LPG ProductionHydrofluoric acidsecondary4.82e-055.39e-06LPG ProductionHydrofluoric acidsecondary4.82e-055.39e-06Fuel Oil #6 Prod.Formaldehydesecondary4.37e-054.84e-06LPG ProductionHydrofluoric acidsecondary3.32e-053.37e-05Fuel Oil #6 Prod.Formaldehydesecondary3.32e-053.38e-06LPG ProductionEthanesecondary3.32e-053.62e-06LPG Productio	Japanese Electric Grid	Nickel	model/secondary	1.18e-04	1.32e-05
US electric gridPM-10model/secondary1.12e-041.25e-05Natural Gas Prod.Vanatiumsecondary1.10e-041.23e-05Fuel Oil #4 Prod.Arsenicsecondary1.08e-041.21e-05Japanese Electric GridBariummodel/secondary9.88e-051.10e-05Monitor/moduleAcetic acidprimary8.27e-059.25e-06Fuel Oil #2 Prod.Nitrogen oxidessecondary7.91e-058.84e-06Fuel Oil #2 Prod.Formaldehydesecondary7.91e-058.84e-06Fuel Oil #4 Prod.PMsecondary7.90e-058.84e-06Fuel Oil #4 Prod.PMsecondary5.87e-056.57e-06Fuel Oil #4 Prod.PMsecondary5.87e-056.57e-06Fuel Oil #4 Prod.PMsecondary5.87e-056.57e-06Fuel Oil #6 Prod.Arsenicsecondary5.87e-056.54e-06Fuel Oil #6 Prod.Zinc (elemental)secondary5.87e-055.44e-06LPG ProductionZinc (elemental)secondary4.82e-055.39e-06Fuel Oil #6 Prod.Formaldehydesecondary4.82e-055.39e-06Fuel Oil #6 Prod.Formaldehydesecondary4.36e-055.21e-06Natural Gas Prod.Formaldehydesecondary3.32e-053.48e-06Fuel Oil #6 Prod.Ethanesecondary3.32e-053.66e-06Fuel Oil #6 Prod.PMsecondary3.12e-053.48e-06Fuel Oil #6 Prod.PMsecond	Fuel Oil #2 Prod.	Sulfur oxides	secondary	1.14e-04	1.27e-05
Natural Gas Prod.Vanadiumsecondary1.10e-041.23e-05Fuel Oil #4 Prod.Arsenicsecondary1.08e-041.21e-05Natural Gas Prod.Hydrochloric acidsecondary9.10e-041.13e-05Japanese Electric GridBariummodel/secondary9.88e-051.10e-05Monitor/moduleAcetic acidprimary8.27e-059.25e-06Fuel Oil #2 Prod.Nitrogen oxidessecondary7.91e-058.84e-06Leu Oil #4 Prod.Formaldehydesecondary7.91e-058.84e-06Fuel Oil #4 Prod.PMsecondary5.87e-056.57e-06Fuel Oil #4 Prod.Arsenicsecondary5.87e-056.57e-06Fuel Oil #6 Prod.Arsenicsecondary5.50e-056.15e-06LPG ProductionZinc (elemental)secondary5.50e-056.15e-06LPG ProductionHydrofluoric acidsecondary4.82e-055.39e-06Natural Gas Prod.Formaldehydesecondary4.37e-054.88e-06Natural Gas Prod.Formaldehydesecondary3.28e-053.34e-06Puel Oil #6 Prod.Formaldehydesecondary3.38e-053.37e-06Puel Oil #0 Prod.Formaldehydesecondary3.28e-053.37e-06Natural Gas Prod.Formaldehydesecondary3.32e-053.37e-06Puel Oil #6 Prod.Phenolsecondary3.32e-053.37e-06Panel componentsHydrochloric acidprimary3.32e-053.37e-06Pa	US electric grid	PM-10	model/secondary	1.12e-04	1.25e-05
Fuel Oil #4 Prod.Arsenicsecondary1.08e-041.21e-05Natural Gas Prod.Hydrochloric acidsecondary1.01e-041.13e-05Japanese Electric GridBariummodel/secondary9.88e-051.10e-05Monitor/moduleAcetic acidprimary8.27e-059.25e-06Fuel Oil #2 Prod.Nitrogen oxidessecondary7.91e-058.85e-06Fuel Oil #4 Prod.Formaldehydesecondary7.90e-058.84e-06Fuel Oil #4 Prod.PMsecondary5.27e-056.47e-06Fuel Oil #4 Prod.PMsecondary5.75e-056.47e-06Fuel Oil #4 Prod.PMsecondary5.75e-056.47e-06Fuel Oil #4 Prod.Zinc (elemental)secondary5.75e-056.47e-06Japanese Electric GridNaphthalenemodel/secondary5.50e-055.21e-06Fuel Oil #6 Prod.Formaldehydesecondary4.82e-055.39e-06Fuel Oil #6 Prod.Formaldehydesecondary4.37e-054.48e-06LPG ProductionHydrofluoric acidsecondary4.37e-054.48e-06LPG ProductionEthanesecondary3.72e-053.48e-06LPG ProductionHydrofluoric acidprimary3.88e-053.73e-06LPG ProductionEthanesecondary3.72e-053.48e-06LPG ProductionHydrofluoric acidprimary3.88e-053.75e-06Panel componentsHydrofluoric acidprimary3.35e-053.52e-06Fuel O	Natural Gas Prod.	Vanadium	secondary	1.10e-04	1.23e-05
Natural Gas Prod.Hydrochloric acidsecondary1.01e-041.13e-05Japanese Electric GridBariummodel/secondary9.88e-051.10e-05Monitor/moduleAcetic acidprimary8.27e-059.25e-06Fuel Oli #2 Prod.Nitrogen oxidessecondary8.20e-059.17e-06Fuel Oli #4 Prod.Formaldehydesecondary7.91e-058.85e-06LPG ProductionMolydenumsecondary7.90e-058.84e-06Fuel Oli #4 Prod.PMsecondary5.87e-056.57e-06LPG ProductionZinc (elemental)secondary5.75e-056.44e-06LPG ProductionZinc (elemental)secondary5.50e-056.15e-06LPG ProductionHydrofluoric acidsecondary4.82e-055.39e-06Fuel Oli #6 Prod.Formaldehydesecondary4.82e-055.21e-06Natirul Gas Prod.Formaldehydesecondary4.37e-054.88e-06LPG ProductionHydrofluoric acidsecondary3.72e-054.46e-05LPG ProductionEthanesecondary3.72e-053.84e-06LPG ProductionPhenolsecondary3.72e-053.84e-06LPG ProductionPhenolsecondary3.72e-053.84e-06LPG ProductionPhenolsecondary3.72e-053.84e-06LPG ProductionPhenolsecondary3.72e-053.84e-06LPG ProductionPhenolsecondary3.15e-053.52e-06Panel componentsHydrofluor	Fuel Oil #4 Prod.	Arsenic	secondary	1.08e-04	1.21e-05
Japanese Electric GridBariummodel/secondary9.88e-051.10e-05Monitor/moduleAcetic acidprimary8.27e-059.25e-06Fuel Oil #2 Prod.Nitrogen oxidessecondary8.20e-059.17e-06Euel Oil #4 Prod.Formaldehydesecondary7.91e-058.85e-06Fuel Oil #4 Prod.Pormaldehydesecondary7.90e-058.84e-06Fuel Oil #4 Prod.PMsecondary6.24e-056.97e-06Fuel Oil #4 Prod.Arsenicsecondary5.75e-056.44e-06LPG ProductionZinc (elemental)secondary5.75e-056.44e-06Japanese Electric GridNaphthalenemodel/secondary5.50e-056.15e-06LPG ProductionHydrofluoric acidsecondary4.82e-055.39e-06Fuel Oil #0 Prod.Formaldehydesecondary4.82e-054.34e-06Natural Gas Prod.Formaldehydesecondary4.37e-054.46e-05LPG ProductionEthanesecondary3.32e-053.84e-06LPG ProductionPhenolsecondary3.32e-053.34e-06LPG ProductionPhenolsecondary3.32e-053.52e-06Fuel Oil #0 Prod.PMsecondary3.12e-053.37e-06Panel componentsHydrochloric acidprimary3.28e-053.52e-06Fuel Oil #0 Prod.PMsecondary3.12e-053.34e-06LPG ProductionPentanesecondary3.12e-053.37e-06Fuel Oil #0 Prod.PM <td>Natural Gas Prod.</td> <td>Hydrochloric acid</td> <td>secondary</td> <td>1.01e-04</td> <td>1.13e-05</td>	Natural Gas Prod.	Hydrochloric acid	secondary	1.01e-04	1.13e-05
Monitor/moduleAcetic acidprimary8.27e-059.25e-06Fuel Oil #2 Prod.Nitrogen oxidessecondary8.20e-059.17e-06Fuel Oil #4 Prod.Formaldehydesecondary7.9le-058.85e-06LPG ProductionMolybdenumsecondary7.9le-058.84e-06Fuel Oil #4 Prod.PMsecondary6.24e-056.97e-06Fuel Oil #4 Prod.Arsenicsecondary5.87e-056.57e-06LPG ProductionZinc (elemental)secondary5.50e-056.15e-06LPG ProductionHydrofluoric acidsecondary4.82e-055.39e-06LPG ProductionHydrofluoric acidsecondary4.82e-055.39e-06Natural Gas Prod.Formaldehydesecondary4.37e-054.48e-06Natural Gas Prod.Formaldehydesecondary3.78e-054.34e-06LPG ProductionEthanesecondary3.72e-054.16e-06LPG ProductionEthanesecondary3.32e-053.54e-06LPG ProductionPhenolsecondary3.32e-053.52e-06LPG ProductionHeptaneprimary3.28e-053.52e-06LPG ProductionPhenolsecondary3.15e-053.52e-06LPG ProductionPhenolsecondary3.15e-053.52e-06LPG ProductionPhenolsecondary3.15e-053.52e-06Fuel Oil #0 Prod.PMsecondary3.15e-053.52e-06Fuel Oil #0 Prod.PMsecondary3.12e-05 <t< td=""><td>Japanese Electric Grid</td><td>Barium</td><td>model/secondary</td><td>9.88e-05</td><td>1.10e-05</td></t<>	Japanese Electric Grid	Barium	model/secondary	9.88e-05	1.10e-05
Fuel Oil #2 Prod.Nitrogen oxidessecondary8.20e-059.17e-06Fuel Oil #4 Prod.Formaldehydesecondary7.91e-058.84e-06LPG ProductionMolybdenumsecondary7.90e-058.84e-06Fuel Oil #4 Prod.PMsecondary6.24e-056.97e-06Fuel Oil #6 Prod.Arsenicsecondary5.87e-056.44e-06Japanese Electric GridNaphthalenemodel/secondary5.50e-056.15e-06LPG ProductionHydrofluoric acidsecondary4.86e-055.39e-06Fuel Oil #6 Prod.Formaldehydesecondary4.66e-055.21e-06Natural Gas Prod.Formaldehydesecondary4.37e-054.88e-06Moltor/moduleAntimonyprimary3.88e-054.34e-06LPG ProductionEthanesecondary3.72e-054.16e-06LPG ProductionEthanesecondary3.35e-053.75e-06Panel componentsHydrochloric acidprimary3.28e-053.66e-06Fuel Oil #0 Prod.PMsecondary3.15e-053.25e-06Panel componentsHeptaneprimary3.28e-053.66e-06Fuel Oil #0 Prod.PMsecondary3.12e-053.60e-06Fuel Oil #0 Prod.PMsecondary3.12e-053.60e-06Fuel Oil #12 Prod.Arsenicsecondary3.12e-053.07e-06Japanese Electric GridOppermodel/secondary2.06e-052.91e-06Japanese Electric GridCopperm	Monitor/module	Acetic acid	primary	8.27e-05	9.25e-06
Fuel Oil #4 Prod.Formaldehydesecondary7.91e-058.85e-06LPG ProductionMolybdenumsecondary7.90e-058.84e-06Fuel Oil #4 Prod.PMsecondary6.24e-056.97e-06Fuel Oil #6 Prod.Arsenicsecondary5.87e-056.57e-06LPG ProductionZinc (elemental)secondary5.75e-056.44e-06Japanese Electric GridNaphthalenemodel/secondary5.50e-056.15e-06LPG ProductionHydrofluoric acidsecondary4.82e-055.39e-06Fuel Oil #6 Prod.Formaldehydesecondary4.37e-054.88e-06Natural Gas Prod.Formaldehydesecondary4.37e-054.48e-06Ded roductionEthanesecondary3.72e-054.16e-06LPG ProductionEthanesecondary3.72e-054.16e-06LPG ProductionPhenolsecondary3.35e-053.75e-06Panel componentsHydrochloric acidprimary3.35e-053.75e-06Panel componentsHydrochloric acidprimary3.35e-053.52e-06LPG ProductionPenanesecondary3.15e-053.49e-06Fuel Oil #6 Prod.PMsecondary3.12e-053.49e-06Panel componentsHydrochloric acidprimary3.12e-053.49e-06Fuel Oil #2 Prod.Arsenicsecondary3.12e-053.49e-06Fuel Oil #2 Prod.Pentanesecondary2.16e-052.19e-06Japanese Electric GridHydro	Fuel Oil #2 Prod.	Nitrogen oxides	secondary	8.20e-05	9.17e-06
LPG ProductionMolybdenumsecondary7.90e-058.84e-06Fuel Oil #4 Prod,PMsecondary6.24e-056.97e-06Fuel Oil #6 Prod,Arsenicsecondary5.87e-056.57e-06LPG ProductionZinc (elemental)secondary5.75e-056.14e-06Japanese Electric GridNaphthalenemodel/secondary5.50e-056.15e-06LPG ProductionHydrofluoric acidsecondary4.82e-055.39e-06Fuel Oil #6 Prod,Formaldehydesecondary4.66e-055.21e-06Natural Gas Prod,Formaldehydesecondary4.37e-054.88e-06Monitor/moduleAntimonyprimary3.88e-054.34e-06LPG ProductionEthanesecondary3.72e-054.16e-06LPG ProductionPhenolsecondary3.72e-053.64e-06Panel componentsHydrochloric acidprimary3.35e-053.75e-06Panel componentsHydrochloric acidprimary3.35e-053.52e-06Fuel Oil #6 Prod,PMsecondary3.12e-053.49e-06Fuel Oil #2 Prod,Pentanesecondary3.12e-053.49e-06Fuel Oil #2 Prod,Arsenicsecondary3.04e-053.49e-06Panel componentsPMprimary2.74e-053.07e-06US eletric GridHydrofluoric acidmodel/secondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.16e-052.48e-06LPG ProductionHexa	Fuel Oil #4 Prod.	Formaldehyde	secondary	7.91e-05	8.85e-06
Fuel Oil #4 Prod.PMsecondary6.24e-056.97e-06Fuel Oil #6 Prod.Arsenicsecondary5.87e-056.57e-06LPG ProductionZinc (elemental)secondary5.75e-056.44e-06Japanese Electric GridNaphthalenemodel/secondary5.50e-056.15e-06LPG ProductionHydrofluoric acidsecondary4.82e-055.39e-06Natural Gas Prod.Formaldehydesecondary4.37e-054.88e-06Monitor/moduleAntimonyprimary3.88e-054.34e-06LPG ProductionEthanesecondary3.72e-054.16e-06LPG ProductionEthanesecondary3.43e-053.84e-06Panel componentsHydrofloric acidprimary3.35e-053.66e-06Panel componentsHeptaneprimary3.32e-053.66e-06Fuel Oil #6 Prod.PMsecondary3.12e-053.52e-06LPG ProductionPentaneprimary3.12e-053.52e-06LPG ProductionPentanesecondary3.12e-053.49e-06Guel Oil #2 Prod.Arsenicsecondary3.12e-053.49e-06Fuel Oil #2 Prod.Arsenicsecondary2.60e-052.91e-06Val Oil #2 Prod.Arsenicsecondary2.60e-052.91e-06Monitor/moduleAcetoneprimary2.22e-052.48e-06Monitor/moduleAcetoneprimary2.16e-052.42e-06Monitor/moduleAcetoneprimary2.16e-052.42e-	LPG Production	Molybdenum	secondary	7.90e-05	8.84e-06
Fuel Oil #6 Prod.Arsenicsecondary5.87e-056.57e-06LPG ProductionZinc (elemental)secondary5.75e-056.44e-06Japanese Electric GridNaphthalenemodel/secondary5.50e-056.15e-06LPG ProductionHydrofluoric acidsecondary4.82e-055.39e-06Fuel Oil #6 Prod.Formaldehydesecondary4.66e-055.21e-06Matural Gas Prod.Formaldehydesecondary4.37e-054.88e-06Monitor/moduleAntimonyprimary3.88e-054.34e-06LPG ProductionEthanesecondary3.72e-054.16e-06LPG ProductionPhenolsecondary3.35e-053.84e-06Panel componentsHydrochloric acidprimary3.35e-053.52e-06Fuel Oil #6 Prod.PMsecondary3.15e-053.52e-06LPG ProductionPentaneprimary3.12e-053.49e-06Fuel Oil #2 Prod.Arsenicsecondary3.12e-053.49e-06Guel Oil #2 Prod.Arsenicsecondary3.12e-053.49e-06LPG ProductionPentaneprimary2.74e-053.07e-06US electric gridHydrofluoric acidmodel/secondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.16e-052.42e-06Monitor/moduleAcetoneprimary2.22e-052.48e-06LPG ProductionHexanesecondary2.16e-052.38e-06 <trr<td>Monitor/moduleCopperp</trr<td>	Fuel Oil #4 Prod.	PM	secondary	6.24e-05	6.97e-06
LPG ProductionZinc (elemental)secondary5.75e-056.44e-06Japanese Electric GridNaphthalenemodel/secondary5.50e-056.15e-06LPG ProductionHydrofluoric acidsecondary4.82e-055.39e-06Fuel Oil #6 Prod.Formaldehydesecondary4.66e-055.21e-06Natural Gas Prod.Formaldehydesecondary4.37e-054.88e-06Monitor/moduleAntimonyprimary3.88e-054.34e-06LPG ProductionEthanesecondary3.72e-054.16e-06LPG ProductionPhenolsecondary3.43e-053.84e-06Panel componentsHydrochloric acidprimary3.35e-053.75e-06Panel componentsHeptaneprimary3.28e-053.66e-06Fuel Oil #6 Prod.PMsecondary3.15e-053.52e-06LPG ProductionPentanesecondary3.12e-053.49e-06LPG ProductionPentanesecondary3.12e-053.49e-06Suel Oil #2 Prod.Arsenicsecondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.16e-052.48e-06LPG ProductionHexanesecondary2.16e-052.30e-06Monitor/moduleAcetoneprimary2.32e-052.48e-06LPG ProductionHexanesecondary2.16e-052.30e-06Monitor/moduleCopperprimary </td <td>Fuel Oil #6 Prod.</td> <td>Arsenic</td> <td>secondary</td> <td>5.87e-05</td> <td>6.57e-06</td>	Fuel Oil #6 Prod.	Arsenic	secondary	5.87e-05	6.57e-06
Japanese Electric GridNaphthalenemodel/secondary5.50e-056.15e-06LPG ProductionHydrofluoric acidsecondary4.82e-055.39e-06Fuel Oil #6 Prod.Formaldehydesecondary4.66e-055.21e-06Natural Gas Prod.Formaldehydesecondary4.37e-054.88e-06Monitor/moduleAntimonyprimary3.88e-054.34e-06LPG ProductionEthanesecondary3.72e-054.16e-06LPG ProductionPhenolsecondary3.43e-053.84e-06Panel componentsHydrochloric acidprimary3.35e-053.75e-06Panel componentsHeptaneprimary3.12e-053.66e-06LPG ProductionPentanesecondary3.12e-053.49e-06LPG ProductionPentanesecondary3.12e-053.49e-06Panel componentsHeptanesecondary3.12e-053.49e-06LPG ProductionPentanesecondary3.12e-053.40e-06LPG ProductionPentanesecondary3.04e-053.40e-06Luel Oil #2 Prod.Arsenicsecondary2.06e-052.91e-06Japanese Electric GridCoppermodel/secondary2.40e-052.69e-06Monitor/moduleAcetoneprimary2.22e-052.48e-06LPG ProductionHexanesecondary2.16e-052.42e-06Natural Gas Prod.Fluorides (F-)secondary2.16e-052.30e-06Monitor/moduleCopperprimary	LPG Production	Zinc (elemental)	secondary	5.75e-05	6.44e-06
LPG ProductionHydrofluoric acidsecondary4.82e-055.39e-06Fuel Oil #6 Prod.Formaldehydesecondary4.66e-055.21e-06Natural Gas Prod.Formaldehydesecondary4.37e-054.88e-06Monitor/moduleAntimonyprimary3.88e-054.34e-06LPG ProductionEthanesecondary3.72e-054.16e-06LPG ProductionPhenolsecondary3.43e-053.84e-06Panel componentsHydrochloric acidprimary3.35e-053.75e-06Panel componentsHeptaneprimary3.28e-053.66e-06Fuel Oil #6 Prod.PMsecondary3.12e-053.52e-06Fuel Oil #6 Prod.PMsecondary3.12e-053.40e-06Fuel Oil #2 Prod.Pentanesecondary3.12e-053.40e-06Fuel Coil #2 Prod.Arsenicsecondary3.04e-053.40e-06Panel componentsPMprimary2.74e-053.07e-06US electric gridHydrofluoric acidmodel/secondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.16e-052.48e-06LPG ProductionHexanesecondary2.16e-052.42e-06Natural Gas Prod.Fluorides (F-)secondary2.16e-052.30e-06Monitor/moduleCopperprimary2.06e-052.30e-06Monitor/moduleCopperprimary2.06e-052.30e-06Monitor/moduleCopperprimary2.06e-05 <td>Japanese Electric Grid</td> <td>Naphthalene</td> <td>model/secondary</td> <td>5.50e-05</td> <td>6.15e-06</td>	Japanese Electric Grid	Naphthalene	model/secondary	5.50e-05	6.15e-06
Fuel Oil #6 Prod.Formaldehydesecondary4.66e-055.21e-06Natural Gas Prod.Formaldehydesecondary4.37e-054.88e-06Monitor/moduleAntimonyprimary3.88e-054.34e-06LPG ProductionEthanesecondary3.72e-054.16e-06LPG ProductionPhenolsecondary3.43e-053.84e-06Panel componentsHydrochloric acidprimary3.35e-053.75e-06Panel componentsHeptaneprimary3.28e-053.66e-06Fuel Oil #6 Prod.PMsecondary3.12e-053.49e-06LPG ProductionPentanesecondary3.12e-053.49e-06Fuel Oil #2 Prod.Arsenicsecondary3.04e-053.40e-06Panel componentsPMprimary2.74e-053.07e-06US electric gridHydrofluoric acidmodel/secondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.16e-052.48e-06LPG ProductionHexanesecondary2.16e-052.48e-06LPG ProductionHexanesecondary2.16e-052.43e-06LPG ProductionHexanesecondary2.13e-052.38e-06Monitor/moduleAcetoneprimary2.22e-052.48e-06Natural Gas Prod.Fluorides (F-)secondary2.16e-052.30e-06Natural Gas Prod.Fluorides (F-)secondary2.16e-052.30e-06Nonitor/moduleCopperprimary2.06e-05 <td>LPG Production</td> <td>Hydrofluoric acid</td> <td>secondary</td> <td>4.82e-05</td> <td>5.39e-06</td>	LPG Production	Hydrofluoric acid	secondary	4.82e-05	5.39e-06
Natural Gas Prod.Formaldehydesecondary4.37e-054.88e-06Monitor/moduleAntimonyprimary3.88e-054.34e-06LPG ProductionEthanesecondary3.72e-054.16e-06LPG ProductionPhenolsecondary3.43e-053.84e-06Panel componentsHydrochloric acidprimary3.35e-053.75e-06Panel componentsHeptaneprimary3.28e-053.66e-06Fuel Oil #6 Prod.PMsecondary3.15e-053.52e-06LPG ProductionPentanesecondary3.12e-053.49e-06Fuel Oil #2 Prod.Arsenicsecondary3.04e-053.40e-06Panel componentsPMprimary2.74e-053.07e-06US electric gridHydrofluoric acidmodel/secondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.40e-052.69e-06Natural Gas Prod.Hexanesecondary2.16e-052.42e-06Natural Gas Prod.Fluorides (F-)secondary2.16e-052.38e-06Natural Gas Prod.Fluorides (F-)primary2.13e-052.38e-06Fuel Oil #2 Prod.Formaldehydesecondary2.06e-052.30e-06Natural Gas Prod.Fluorides (F-)secondary2.13e-052.38e-06Monitor/moduleCopperprimary2.06e-052.30e-06Fuel Oil #2 Prod.Formaldehydesecondary2.06e-052.30e-06Monitor/modulePolychlorinated biphen	Fuel Oil #6 Prod.	Formaldehyde	secondary	4.66e-05	5.21e-06
Monitor/moduleAntimonyprimary3.88e-054.34e-06LPG ProductionEthanesecondary3.72e-054.16e-06LPG ProductionPhenolsecondary3.43e-053.84e-06Panel componentsHydrochloric acidprimary3.35e-053.75e-06Panel componentsHeptaneprimary3.28e-053.66e-06Fuel Oil #6 Prod.PMsecondary3.15e-053.52e-06LPG ProductionPentanesecondary3.12e-053.49e-06Fuel Oil #2 Prod.Arsenicsecondary3.04e-053.40e-06Panel componentsPMprimary2.74e-053.07e-06US electric gridHydrofluoric acidmodel/secondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.40e-052.69e-06LPG ProductionHexanesecondary2.16e-052.42e-06Natural Gas Prod.Fluorides (F-)secondary2.13e-052.38e-06Monitor/moduleCopperprimary2.06e-052.30e-06Fuel Oil #2 Prod.Formaldehydesecondary2.13e-052.30e-06Nonitor/moduleCopperprimary2.06e-052.30e-06Monitor/modulePolychlorinated biphenylsprimary1.94e-052.30e-06Monitor/modulePolychlorinated biphenylsprimary1.94e-052.17e-06	Natural Gas Prod.	Formaldehyde	secondary	4.37e-05	4.88e-06
LPG ProductionEthanesecondary3.72e-054.16e-06LPG ProductionPhenolsecondary3.43e-053.84e-06Panel componentsHydrochloric acidprimary3.35e-053.75e-06Panel componentsHeptaneprimary3.28e-053.66e-06Fuel Oil #6 Prod.PMsecondary3.15e-053.52e-06LPG ProductionPentanesecondary3.12e-053.49e-06Fuel Oil #2 Prod.Arsenicsecondary3.04e-053.40e-06Panel componentsPMprimary2.74e-053.07e-06US electric gridHydrofluoric acidmodel/secondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.40e-052.69e-06LPG ProductionHexanesecondary2.16e-052.42e-06Natural Gas Prod.Fluorides (F-)secondary2.13e-052.38e-06Monitor/moduleCopperprimary2.06e-052.30e-06Fuel Oil #2 Prod.Formaldehydesecondary2.13e-052.30e-06Monitor/moduleCopperprimary2.06e-052.30e-06Fuel Oil #2 Prod.Formaldehydesecondary2.06e-052.30e-06Monitor/modulePolychlorinated biphenylsprimary1.94e-052.17e-06	Monitor/module	Antimony	primary	3.88e-05	4.34e-06
LPG ProductionPhenolsecondary3.43e-053.84e-06Panel componentsHydrochloric acidprimary3.35e-053.75e-06Panel componentsHeptaneprimary3.28e-053.66e-06Fuel Oil #6 Prod.PMsecondary3.15e-053.52e-06LPG ProductionPentanesecondary3.12e-053.49e-06Fuel Oil #2 Prod.Arsenicsecondary3.04e-053.40e-06Panel componentsPMprimary2.74e-053.07e-06US electric gridHydrofluoric acidmodel/secondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.40e-052.69e-06Monitor/moduleAcetoneprimary2.22e-052.48e-06LPG ProductionHexanesecondary2.16e-052.42e-06Natural Gas Prod.Fluorides (F-)secondary2.13e-052.38e-06Monitor/moduleCopperprimary2.06e-052.30e-06Fuel Oil #2 Prod.Formaldehydesecondary2.06e-052.30e-06Monitor/moduleCopperprimary2.06e-052.30e-06Monitor/modulePolychlorinated biphenylsprimary1.94e-052.17e-06	LPG Production	Ethane	secondary	3.72e-05	4.16e-06
Panel componentsHydrochloric acidprimary3.35e-053.75e-06Panel componentsHeptaneprimary3.28e-053.66e-06Fuel Oil #6 Prod.PMsecondary3.15e-053.52e-06LPG ProductionPentanesecondary3.12e-053.49e-06Fuel Oil #2 Prod.Arsenicsecondary3.04e-053.40e-06Panel componentsPMprimary2.74e-053.07e-06US electric gridHydrofluoric acidmodel/secondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.22e-052.48e-06LPG ProductionHexanesecondary2.16e-052.42e-06Natural Gas Prod.Fluorides (F-)secondary2.13e-052.38e-06Monitor/moduleCopperprimary2.06e-052.30e-06Monitor/moduleCopperprimary2.06e-052.30e-06Monitor/moduleCopperprimary2.06e-052.30e-06Monitor/moduleCopperprimary2.06e-052.30e-06Monitor/modulePolychlorinated biphenylsprimary1.94e-052.17e-06	LPG Production	Phenol	secondary	3.43e-05	3.84e-06
Panel componentsHeptaneprimary3.28e-053.66e-06Fuel Oil #6 Prod.PMsecondary3.15e-053.52e-06LPG ProductionPentanesecondary3.12e-053.49e-06Fuel Oil #2 Prod.Arsenicsecondary3.04e-053.40e-06Panel componentsPMprimary2.74e-053.07e-06US electric gridHydrofluoric acidmodel/secondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.40e-052.69e-06Monitor/moduleAcetoneprimary2.22e-052.48e-06LPG ProductionHexanesecondary2.16e-052.42e-06Natural Gas Prod.Fluorides (F-)secondary2.13e-052.38e-06Monitor/moduleCopperprimary2.06e-052.30e-06Fuel Oil #2 Prod.Formaldehydesecondary2.06e-052.30e-06Monitor/moduleCopperprimary1.94e-052.30e-06	Panel components	Hydrochloric acid	primary	3.35e-05	3.75e-06
Fuel Oil #6 Prod.PMsecondary3.15e-053.52e-06LPG ProductionPentanesecondary3.12e-053.49e-06Fuel Oil #2 Prod.Arsenicsecondary3.04e-053.40e-06Panel componentsPMprimary2.74e-053.07e-06US electric gridHydrofluoric acidmodel/secondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.40e-052.69e-06Monitor/moduleAcetoneprimary2.22e-052.48e-06LPG ProductionHexanesecondary2.16e-052.42e-06Natural Gas Prod.Fluorides (F-)secondary2.13e-052.38e-06Monitor/moduleCopperprimary2.06e-052.30e-06Fuel Oil #2 Prod.Formaldehydesecondary2.06e-052.30e-06Fuel Oil #2 Prod.Polychlorinated biphenylsprimary1.94e-052.17e-06	Panel components	Heptane	primary	3.28e-05	3.66e-06
LPG ProductionPentanesecondary3.12e-053.49e-06Fuel Oil #2 Prod.Arsenicsecondary3.04e-053.40e-06Panel componentsPMprimary2.74e-053.07e-06US electric gridHydrofluoric acidmodel/secondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.40e-052.69e-06Monitor/moduleAcetoneprimary2.22e-052.48e-06LPG ProductionHexanesecondary2.16e-052.42e-06Natural Gas Prod.Fluorides (F-)secondary2.13e-052.38e-06Monitor/moduleCopperprimary2.06e-052.30e-06Fuel Oil #2 Prod.Formaldehydesecondary2.06e-052.30e-06Fuel Oil #2 Prod.Polychlorinated biphenylsprimary1.94e-052.17e-06	Fuel Oil #6 Prod.	PM	secondary	3.15e-05	3.52e-06
Fuel Oil #2 Prod.Arsenicsecondary3.04e-053.40e-06Panel componentsPMprimary2.74e-053.07e-06US electric gridHydrofluoric acidmodel/secondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.40e-052.69e-06Monitor/moduleAcetoneprimary2.22e-052.48e-06LPG ProductionHexanesecondary2.16e-052.42e-06Natural Gas Prod.Fluorides (F-)secondary2.13e-052.38e-06Monitor/moduleCopperprimary2.06e-052.30e-06Fuel Oil #2 Prod.Formaldehydesecondary2.06e-052.30e-06Monitor/modulePolychlorinated biphenylsprimary1.94e-052.17e-06	LPG Production	Pentane	secondary	3.12e-05	3.49e-06
Panel componentsPMprimary2.74e-053.07e-06US electric gridHydrofluoric acidmodel/secondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.40e-052.69e-06Monitor/moduleAcetoneprimary2.22e-052.48e-06LPG ProductionHexanesecondary2.16e-052.42e-06Natural Gas Prod.Fluorides (F-)secondary2.13e-052.38e-06Monitor/moduleCopperprimary2.06e-052.30e-06Fuel Oil #2 Prod.Formaldehydesecondary2.06e-052.30e-06Monitor/modulePolychlorinated biphenylsprimary1.94e-052.17e-06	Fuel Oil #2 Prod.	Arsenic	secondary	3.04e-05	3.40e-06
US electric gridHydrofluoric acidmodel/secondary2.60e-052.91e-06Japanese Electric GridCoppermodel/secondary2.40e-052.69e-06Monitor/moduleAcetoneprimary2.22e-052.48e-06LPG ProductionHexanesecondary2.16e-052.42e-06Natural Gas Prod.Fluorides (F-)secondary2.13e-052.38e-06Monitor/moduleCopperprimary2.06e-052.30e-06Fuel Oil #2 Prod.Formaldehydesecondary2.06e-052.30e-06Monitor/modulePolychlorinated biphenylsprimary1.94e-052.17e-06	Panel components	PM	primary	2.74e-05	3.07e-06
Japanese Electric GridCoppermodel/secondary2.40e-052.69e-06Monitor/moduleAcetoneprimary2.22e-052.48e-06LPG ProductionHexanesecondary2.16e-052.42e-06Natural Gas Prod.Fluorides (F-)secondary2.13e-052.38e-06Monitor/moduleCopperprimary2.06e-052.30e-06Fuel Oil #2 Prod.Formaldehydesecondary2.06e-052.30e-06Monitor/modulePolychlorinated biphenylsprimary1.94e-052.17e-06	US electric grid	Hydrofluoric acid	model/secondary	2.60e-05	2.91e-06
Monitor/moduleAcetoneprimary2.22e-052.48e-06LPG ProductionHexanesecondary2.16e-052.42e-06Natural Gas Prod.Fluorides (F-)secondary2.13e-052.38e-06Monitor/moduleCopperprimary2.06e-052.30e-06Fuel Oil #2 Prod.Formaldehydesecondary2.06e-052.30e-06Monitor/modulePolychlorinated biphenylsprimary1.94e-052.17e-06	Japanese Electric Grid	Copper	model/secondary	2.40e-05	2.69e-06
LPG ProductionHexanesecondary2.16e-052.42e-06Natural Gas Prod.Fluorides (F-)secondary2.13e-052.38e-06Monitor/moduleCopperprimary2.06e-052.30e-06Fuel Oil #2 Prod.Formaldehydesecondary2.06e-052.30e-06Monitor/modulePolychlorinated biphenylsprimary1.94e-052.17e-06	Monitor/module	Acetone	primary	2.22e-05	2.48e-06
Natural Gas Prod.Fluorides (F-)secondary2.13e-052.38e-06Monitor/moduleCopperprimary2.06e-052.30e-06Fuel Oil #2 Prod.Formaldehydesecondary2.06e-052.30e-06Monitor/modulePolychlorinated biphenylsprimary1.94e-052.17e-06	LPG Production	Hexane	secondary	2.16e-05	2.42e-06
Monitor/moduleCopperprimary2.06e-052.30e-06Fuel Oil #2 Prod.Formaldehydesecondary2.06e-052.30e-06Monitor/modulePolychlorinated biphenylsprimary1.94e-052.17e-06	Natural Gas Prod.	Fluorides (F-)	secondary	2.13e-05	2.38e-06
Fuel Oil #2 Prod.Formaldehydesecondary2.06e-052.30e-06Monitor/modulePolychlorinated biphenylsprimary1.94e-052.17e-06	Monitor/module	Copper	primarv	2.06e-05	2.30e-06
Monitor/module Polychlorinated biphenyls primary 1.94e-05 2.17e-06	Fuel Oil #2 Prod.	Formaldehyde	secondary	2.06e-05	2.30e-06
	Monitor/module	Polychlorinated biphenyls	primary	1.94e-05	2.17e-06

 Table M-40. LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Selenium	secondary	1.91e-05	2.14e-06
Fuel Oil #4 Prod.	Hydrochloric acid	secondary	1.90e-05	2.12e-06
US electric grid	Formaldehyde	model/secondary	1.88e-05	2.10e-06
Japanese Electric Grid	2-Chloroacetophenone	model/secondary	1.85e-05	2.07e-06
Fuel Oil #2 Prod.	PM	secondary	1.85e-05	2.07e-06
Japanese Electric Grid	Bromomethane	model/secondary	1.83e-05	2.05e-06
Natural Gas Prod.	Nitrous oxide	secondary	1.82e-05	2.04e-06
LPG Production	Nickel	secondary	1.39e-05	1.56e-06
US electric grid	Benzene	model/secondary	1.35e-05	1.51e-06
Natural Gas Prod.	Zinc (elemental)	secondary	1.34e-05	1.50e-06
Natural Gas Prod.	Ethane	secondary	1.10e-05	1.23e-06
Monitor/module	PM	primary	1.10e-05	1.23e-06
LPG Production	PM-10	secondary	1.08e-05	1.21e-06
LPG Production	Antimony	secondary	1.03e-05	1.15e-06
Fuel Oil #6 Prod.	Hydrochloric acid	secondary	1.01e-05	1.13e-06
Japanese Electric Grid	Cyanide (-1)	model/secondary	9.61e-06	1.07e-06
Natural Gas Prod.	Pentane	secondary	9.24e-06	1.03e-06
Panel components	Toluene	primary	9.09e-06	1.02e-06
Fuel Oil #4 Prod.	Nitrous oxide	secondary	8.86e-06	9.91e-07
Monitor/module	Chromium	primary	8.84e-06	9.88e-07
US electric grid	Phosphorus (yellow or white)	model/secondary	7.61e-06	8.51e-07
Japanese Electric Grid	2,3,7,8-TCDD	model/secondary	7.47e-06	8.36e-07
LPG Production	Aluminum (+3)	secondary	6.79e-06	7.60e-07
US electric grid	Nitrous oxide	model/secondary	6.58e-06	7.36e-07
Natural Gas Prod.	Hexane	secondary	6.40e-06	7.15e-07
Monitor/module	Boron	primary	6.19e-06	6.93e-07
Monitor/module	Lead	primary	6.17e-06	6.90e-07
Fuel Oil #4 Prod.	Phosphorus (vellow or white)	secondary	5.71e-06	6.38e-07
Fuel Oil #2 Prod.	Hydrochloric acid	secondary	5.39e-06	6.02e-07
LCD glass mfg.	PM	primary	5.06e-06	5.66e-07
Fuel Oil #6 Prod.	Nitrous oxide	secondary	4.94e-06	5.53e-07
LPG Production	Chromium (VI)	secondary	4.84e-06	5.41e-07
Japanese Electric Grid	Cobalt	model/secondary	3.98e-06	4.45e-07
Fuel Oil #4 Prod.	Selenium	secondary	3.85e-06	4.31e-07
Fuel Oil #4 Prod.	Fluorides (F-)	secondary	3.79e-06	4.24e-07
Monitor/module	Cvanide (-1)	primary	3.66e-06	4.10e-07
Natural Gas Prod.	Molybdenum	secondary	3.64e-06	4.08e-07
US electric grid	Zinc (elemental)	model/secondary	3 51e-06	3 93e-07
Japanese Electric Grid	Carbon disulfide	model/secondary	3.43e-06	3.84e-07
Fuel Oil #6 Prod	Phosphorus (vellow or white)	secondary	3 33e-06	3 72e-07
LPG Production	Nitrate	secondary	2.91e-06	3 25e-07
US electric grid	Antimony	model/secondary	2.91e 00	3.21e-07
Natural Gas Prod	Phosphorus (vellow or white)	secondary	2.37e-06	3.12e-07
Natural Gas Prod	Hydrofluoric acid	secondary	2.75e-06	3.07e-07
Japanese Electric Grid	Benzyl chloride	model/secondary	2.75e 00 2.69e-06	3.01e-07
I CD glass mfg	Carbon monoxide	nrimary	2.09e-06	2 89e-07
Evel Oil #2 Prod	Nitrous oxide	secondary	2.590-00	2.090-07
I CD alass mfa	Sulfur oxides	nrimary	2.425-00	2.710-07
Monitor/module	Cyclohevane	primary	2.550-00	2.050-07
Racklight	Diethyl ether	primary	2.226-00	2.400-07
Japanaga Elastria Grid	Chromium (VI)	primary model/seconder-	2.200-06	2.4/e-0/
Japanese Electric Grid	Chroinium (VI)	inodel/secondary	2.15e-06	2.40e-07

Japanese Electric Grid Chloroform model/secondary 2.09-06 2.31-07 ECD Gliß Prod. Sacondary 2.01-06 2.31-07 ECD glass mfg. Lead primary 2.01-06 2.32-07 Fuel Oli M Prod. Phordise (Fs) secondary 1.88-06 2.31-07 EVG Production Copper secondary 1.88-06 1.77-07 EVG Production Chloroscetophrone secondary 1.58-06 1.77-07 EVG Production Diromorehane secondary 1.58-06 1.71-07 EVG Production Diromorehane secondary 1.50-06 1.68-07 Montorimodul Hespanetyldishizane primary 1.32-06 1.58-07 Pagenese Electric Grid Acroscen model/secondary 1.16-06 1.58-07 Pagenese Electric Grid Acroscen model/secondary 1.09-06 1.22-07 Fuel Oli # 2 Prod. Appthalene secondary 1.09-06 1.22-07 Fuel Oli # 2 Prod. Modylofenun secondary 1.09-06 1.22-07 <th>Process Group</th> <th>Material</th> <th>LCI Data Type</th> <th>Terrestrial Toxicity (tox-kg)</th> <th>% of Total</th>	Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Facl Olif & Prod. Sclenium scondary 2.07-06 2.31-07 ICD glass m/g. Icad primary 2.01-06 2.252-07 Fuel Olif & Prod. Hydrogen sulfide scondary 1.83-06 2.210-07 Fuel Olif & Prod. Ommoria scondary 1.83-06 2.175-07 Fuel Olif & Prod. Anmoria scondary 1.54-06 1.75-07 ICO Production Denterhy Hearzanthracene scondary 1.54-06 1.75-07 IPG Production Domesthane scondary 1.54-06 1.71-07 IPG Production Nontormodule Hexamethyldisilizane primary 1.50-06 1.53-07 IPG Production Naptares Electric Grid Acrolein scondary 1.10-06 1.23-07 Japares Electric Grid Acrolein scondary 1.10-06 1.22-07 Fuel Olif & Prod. Hydrogen sulfide scondary 1.00-06 1.22-07 Fuel Olif & Prod. Hydrogen sulfide scondary 9.61-07 1.07-07 Fuel Olif & Prod. Hydrogen	Japanese Electric Grid	Chloroform	model/secondary	2.09e-06	2.34e-07
LCD gas mfg. Lead primary 2.01e.06 2.23e.07 Fuel Oil # Pred. Hydrogen sulfide secondary 1.83e.06 2.20e.07 Fuel Oil # Pred. Gymen sulfide secondary 1.87e.07 2.88e.06 2.18e.07 FUE Or Production Copper secondary 1.87e.07 1.87e.07 1.87e.07 FUE Or Production Dimethy Benzanthracene secondary 1.51e.06 1.77e.07 FUE Or Production Romomerhane secondary 1.53e.06 1.77e.07 Fuel Oil # 2 Prod. Prosphorus (vellow or white) secondary 1.59e.06 1.53e.07 Japanese Electric Grid Acrolein model/secondary 1.19e.06 1.22e.07 Japanese Electric Grid Acrolein secondary 1.09e.06 1.22e.07 Fuel Oil # 2 Prod. Floorides (F.) secondary 1.09e.06 1.22e.07 Fuel Oil # 2 Prod. Floorides (F.) secondary 1.09e.06 1.22e.07 Fuel Oil # 2 Prod. Floorides (F.) secondary 1.09e.06 1.22e.07	Fuel Oil #6 Prod.	Selenium	secondary	2.07e-06	2.31e-07
Fuel Oil #0 Prod.Fluorides (F-)secondary1.99e-062.22-er7Fuel Oil #0 Prod.Hydrogen sulfidesecondary1.67e-061.87e-07Fuel Oil #1 Prod.Ammoniasecondary1.54e-061.77e-17LFO ProductionO Echforoacetophenonesecondary1.54e-061.77e-17LFO ProductionDimethylbenzanturacenesecondary1.54e-061.771e-07Fuel Oil #2 Prod.Phosphorus (yellow or white)secondary1.54e-061.771e-07Fuel Oil #2 Prod.Phosphorus (yellow or white)secondary1.54e-061.86e-07Fuel Oil #2 Prod.Naphtalenesecondary1.12e-061.86e-07Japanese Electric GridAcroleinmodel'secondary1.19e-061.22e-07Fuel Oil #2 Prod.Fluorides (F-)secondary1.09e-061.22e-07Fuel Oil #2 Prod.Fluorides (F)secondary1.09e-061.22e-07Fuel Oil #2 Prod.Fluorides (F)secondary1.09e-061.22e-07Fuel Oil #2 Prod.Fluorides (F)secondary1.09e-061.22e-07Fuel Oil #2 Prod.Fluorides (F)secondary1.09e-061.22e-07Fuel Oil #2 Prod.Hydrogen sulfidesecondary9.61e-071.07e-07Fuel Oil #2 Prod.Molybedrumsecondary9.61e-071.07e-07Fuel Oil #2 Prod.Molybedrumsecondary9.01e-071.01e-07Fuel Oil #2 Prod.Molybedrumsecondary9.01e-071.01e-07Fuel Oil #4 Prod.Mol	LCD glass mfg.	Lead	primary	2.01e-06	2.25e-07
Fuel Oil #4 Prod. Hydragen sulfide secondary 1.83-e06 2.04-c07 LPG Production Copper secondary 1.67e-67 1.76e-07 LPG Production Dimchylbenznthracene secondary 1.54e-06 1.77e-07 LPG Production Dimchylbenznthracene secondary 1.55e-06 1.71e-07 LPG Production Bromomehane secondary 1.55e-06 1.71e-07 Pacl Oil #2 Prod. Phosphorus (ycllow or white) secondary 1.37e-06 1.53e-07 Pard Oil #2 Prod. Hoursing (Sclow or white) secondary 1.24e-06 1.53e-07 Japanese Electric Grid Arcolein model-secondary 1.99e-06 1.22e-07 Fuel Oil #2 Prod. Selonin secondary 1.09e-06 1.22e-07 Fuel Oil #2 Prod. Seloninm secondary 1.09e-06 1.22e-07 Fuel Oil #2 Prod. Seloninm secondary 1.08e-07 1.07e-07 Fuel Oil #4 Prod. Molybemam secondary 9.06e-07 1.07e-07 Fuel Oil #4 Prod. Molybemam </td <td>Fuel Oil #6 Prod.</td> <td>Fluorides (F-)</td> <td>secondary</td> <td>1.99e-06</td> <td>2.22e-07</td>	Fuel Oil #6 Prod.	Fluorides (F-)	secondary	1.99e-06	2.22e-07
LPG Production Copper secondary 1.67:e07 Fuel Oil #4 Prod. Annnonia secondary 1.58:e06 1.76:e07 LPG Production Dimeltylbenzambracene secondary 1.53:e06 1.71:e07 LPG Production Dimentylbenzambracene secondary 1.53:e06 1.71:e07 Fuel Oil #2 Prod. Phosphorus (yellow or white) secondary 1.24:e06 1.58:e07 Monitorimodule Hexamethyldisilizane primary 1.37:e06 1.58:e07 Japanese Electric Grid Aerolein model/secondary 1.11:e06 1.23:e07 Fuel Oil #2 Prod. Fluorids: (F-) secondary 1.09:e06 1.22:e07 Fuel Oil #0 Prod. Hyorigen sulfide secondary 1.09:e06 1.22:e07 Fuel Oil #1 Prod. Selenium secondary 1.09:e06 1.22:e07 Fuel Oil #1 Prod. Molybdenum secondary 9.20:e07 1.06:e07 Fuel Oil #1 Prod. Molybdenum secondary 9.20:e07 1.08:e07 Fuel Oil #1 Prod. Molybdenum seconda	Fuel Oil #4 Prod.	Hydrogen sulfide	secondary	1.83e-06	2.04e-07
Fuel Oil 4 Prod. Armonia secondary 1.58-06 1.78-07 LPG Production Dimethylbenzanthracene secondary 1.54-06 1.716-07 LPG Production Bromomethane secondary 1.55-06 1.716-07 LPG Production Bromomethane secondary 1.55-06 1.716-07 Pacel Oil 2 Prod. Phosphons (yellow or white) secondary 1.26-06 1.58-07 LPG Production Naphthalene secondary 1.26-06 1.28-07 Japanese Electric Grid Acrolein model/secondary 1.09-06 1.22e-07 Fuel Oil # 2 Prod. Selenium secondary 1.09-06 1.22e-07 Fuel Oil # 6 Prod. Hydrogen sulfide secondary 1.09-06 1.22e-07 Fuel Oil # 7 Prod. Majyddenum secondary 9.06-07 1.07e-07 Fuel Oil # 7 Prod. Maydydenum secondary 9.06-07 1.07e-07 Fuel Oil # 7 Prod. Maydydenum secondary 9.06-07 1.07e-07 Fuel Oil # 7 Prod. Molybdenum second	LPG Production	Copper	secondary	1.67e-06	1.87e-07
LPG Production 2-Chlorosectophenone secondary 1.53-e06 1.73-e07 LPG Production Dimethylbenznuhracene secondary 1.53-e06 1.71-e07 LPG Production Bronomethane secondary 1.50-e06 1.68-e07 Fuel Oil #2 Prod. Phosphorus (cellow or white) secondary 1.37-e06 1.58-e07 LPG Production Naphthalene secondary 1.31-e06 1.58-e07 Japanese Electric Grid Acrobin model/secondary 1.09-e06 1.22-e07 Patel Oil #2 Prod. Floridiks (F-) secondary 1.09-e06 1.22-e07 Fuel Oil #2 Prod. Release secondary 1.09-e06 1.22-e07 Fuel Oil #0 Prod. Hydrogen sulfide secondary 1.08-e06 1.22-e07 Fuel Oil #0 Prod. Hydrogen sulfide secondary 1.08-e06 1.20-e07 Fuel Oil #0 Prod. Hydrogen sulfide secondary 9.61-e07 1.07-e07 Fuel Oil #4 Prod. Molybdenum model/secondary 9.23-e07 1.04-e07 Monitorimodule	Fuel Oil #4 Prod.	Ammonia	secondary	1.58e-06	1.76e-07
LPG Production Dimethylbenzanthracene secondary 1.53e-06 1.71e-07 LPG Production Bromomethane secondary 1.58e-06 1.71e-07 Ped Oll #2 Prod. Phosphorus (yellow or white) secondary 1.37e-06 1.53e-07 LPG Production Naphtrahene secondary 1.37e-06 1.53e-07 LPG Production Naphtrahene secondary 1.09e-06 1.22e-07 Fuel Oll #2 Prod. Selenium secondary 1.09e-06 1.22e-07 Fuel Oll #2 Prod. Molybdemm secondary 0.10e-07 1.07e-06 1.20e-07 Fuel Oll #4 Prod. Molybdemm secondary 9.03e-07 1.04e-07 Japanese Electric Grid Methyl ehloride model/secondary 8.3e-07 1.04e-07 Japanese Electric Grid <td< td=""><td>LPG Production</td><td>2-Chloroacetophenone</td><td>secondary</td><td>1.54e-06</td><td>1.73e-07</td></td<>	LPG Production	2-Chloroacetophenone	secondary	1.54e-06	1.73e-07
LPG ProductionBromonchancsecondary1.53e-061.71e-07Fuel Oli #2 Prod.Phosphorus (vellow or white)secondary1.50e-061.68e-07Monitor/moduleHexamehyldisitzneprimary1.37e-061.53e-07Japanese Electric GridAcroleinmodel/secondary1.24e-061.38e-07Japanese Electric GridFlourides (F-)secondary1.09e-061.22e-07Monitor/moduleChromium (VI)primary1.09e-061.22e-07Fuel Oli #2 Prod.Seleniumsecondary1.08e-061.22e-07Fuel Oli #0 Prod.Hydrogen sulfidesecondary1.08e-061.22e-07LPG ProductionMagnaesesecondary1.07e-061.20e-07LPG OroductionMagnaesesecondary9.61e-071.07e-07LPG ProductionMolybdenummodel/secondary9.29e-071.04e-07Monitor/moduleDiethylene glycolprimary9.23e-071.00e-07LPG ProductionBariumsecondary8.35e-079.34e-08LPG ProductionBariumsecondary8.35e-078.94e-07LPG ProductionClaronectophenonemodel/secondary8.03e-078.98e-08LPG ProductionSiliconsecondary8.03e-078.98e-08LPG ProductionClaronectophenonemodel/secondary8.03e-078.98e-08LPG ProductionClaronectophenonemodel/secondary8.03e-078.98e-08LPG ProductionBarium enpd5secondary5.08e-07	LPG Production	Dimethylbenzanthracene	secondary	1.53e-06	1.71e-07
Fuel Oil #2 Prod. Phosphorus (yellow or white) secondary 1.50-60 1.68-67 Monitor/module Hexamethyläsilizane primary 1.37e-66 1.53e-07 Japanese Electric Grid Acrolein model/secondary 1.11e-66 1.23e-07 Fuel Oil #2 Prod. Fluorides (F-) secondary 1.09e-66 1.22e-07 Fuel Oil #2 Prod. Selenium secondary 1.09e-66 1.22e-07 Fuel Oil #2 Prod. Selenium secondary 1.09e-66 1.22e-07 Fuel Oil #2 Prod. Hydrogen sulfide secondary 1.07e-06 1.22e-07 Fuel Oil #4 Prod. Molybdenum model/secondary 9.29e-07 1.04e-07 Monitor/module Diethylene glycol primary 9.23e-07 1.03e-07 Japanese Electric Grid Methyl chloride model/secondary 8.94e-07 1.00e-07 US electric grid 2.Chloroacetophenone model/secondary 8.94e-07 8.94e-08 US electric grid Toluene model/secondary 8.94e-07 8.94e-08 LPG Production<	LPG Production	Bromomethane	secondary	1.53e-06	1.71e-07
Monitor/moduleHexamethyldisilizaneprimary1.37e-061.53e-07LPG ProductionNaphthalenesecondary1.24e-061.38e-07Japanese Electric GridAcroleinmodel/secondary1.09e-061.22e-07Fuel Oli #2 Prod.Chromium (V)primary1.09e-061.22e-07Fuel Oli #2 Prod.Scleniumsecondary1.09e-061.22e-07Fuel Oli #2 Prod.Scleniumsecondary1.09e-061.22e-07Fuel Oli #2 Prod.Scleniumsecondary1.08e-061.21e-07LPG ProductionManganesesecondary1.07e-061.20e-07US electric gridMolybdenummodel/secondary9.61e-071.07e-06Japanese Electric GridMelybdenumsecondary9.23e-071.04e-07Monitor/moduleDiethylene glycolprimary9.23e-071.04e-07JCP ProductionBariumsecondary9.03e-071.00e-07LPG ProductionBariumsecondary8.35e-079.34e-08US electric gridDromomethanemodel/secondary8.35e-079.34e-08LPG ProductionSiliconsecondary8.12e-079.84e-08LPG ProductionCyanide (-1)secondary8.35e-078.98e-08LPG ProductionCyanide (-1)secondary8.35e-078.98e-08LPG ProductionCyanide (-1)secondary6.85e-077.84e-08Japanese Electric GridToluenemodel/secondary6.85e-077.84e-08LPG Produc	Fuel Oil #2 Prod.	Phosphorus (yellow or white)	secondary	1.50e-06	1.68e-07
LPG ProductionNaphthalenesecondary1.24e-061.38e-07Japanese Electric GridAcroleinmodel/secondary1.10e-061.22e-07Monitor/moduleChromium (VI)primary1.09e-061.22e-07Fuel Oli # 2 Prod.Seleniumsecondary1.09e-061.22e-07Leu Oli # 6 Prod.Hydrogen sulfidesecondary1.09e-061.22e-07Leu Oli # 6 Prod.Hydrogen sulfidesecondary1.07e-061.20e-07LPG ProductionManganesesecondary9.01e-071.07e-07Leu Oli # 6 Prod.Molybelnummodel/secondary9.03e-071.04e-07Monitor/moduleDiethy lene glycolprimary9.23e-071.03e-07Japanese Electric GridMethyl chloridemodel/secondary9.03e-071.00e-07LPG ProductionBariumsecondary8.34e-078.94e-08US electric grid2.Chloroactophenonemodel/secondary8.32e-079.08e-08LPG ProductionSiliconsecondary8.04e-078.89e-08LPG ProductionCyanide (1)secondary6.85e-077.84e-08LPG ProductionCyanide (-1)secondary6.85e-077.31e-08LPG ProductionCyanide (-1)secondary6.85e-077.84e-08LPG ProductionCyanide (-1)secondary6.85e-077.84e-08LPG ProductionCyanide (-1)secondary6.85e-077.51e-08LPG ProductionCyanide (-1)secondary6.85e-077.51e-08 <td>Monitor/module</td> <td>Hexamethyldisilizane</td> <td>primary</td> <td>1.37e-06</td> <td>1.53e-07</td>	Monitor/module	Hexamethyldisilizane	primary	1.37e-06	1.53e-07
Japanese Electric GridAcroleinmodel/secondary1.11e-061.25e-07Fuel Oil #2 Prod.Fluorides (F-)secondary1.09e-061.22e-07Fuel Oil #2 Prod.Scleniumsecondary1.09e-061.22e-07Fuel Oil #2 Prod.Scleniumsecondary1.07e-061.22e-07Fuel Oil #6 Prod.Hydrogen sulfidesecondary1.07e-061.20e-07US electric gridMolybdenummodel/secondary9.61e-071.07e-07Fuel Oil #4 Prod.Molybdenumsecondary9.23e-071.04e-07Japanese Electric GridMethyl chloridemodel/secondary9.33e-071.01e-07Japanese Electric GridMethyl chloridemodel/secondary8.34e-071.00e-07Jos electric grid2.Chloroacetophenonemodel/secondary8.37e-079.24e-08US electric grid2.Chloroacetophenonemodel/secondary8.27e-079.24e-08LPG ProductionSiliconsecondary8.12e-079.8e-08LPG ProductionCyanide (-1)secondary8.04e-078.99e-08LPG ProductionCyanide (-1)secondary7.02e-078.58e-08LPG ProductionCyanide (-1)secondary7.02e-077.84e-08Fuel Oil #6 Prod.Ammoniasecondary5.30e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary5.30e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary5.30e-075.57e-08Fuel Oil #7 Prod.<	LPG Production	Naphthalene	secondary	1.24e-06	1.38e-07
Fuel Oil #2 Prod.Fluorides (F-)secondary1.09e-061.22e-07Monitor/moduleChromium (VI)primary1.09e-061.22e-07Fuel Oil #7 Prod.Seleniumsecondary1.09e-061.22e-07Fuel Oil #6 Prod.Hydrogen sulfidesecondary1.07e-061.22e-07US electric gridMolybdenummodel/secondary9.261e-071.07e-06US electric gridMolybdenumsecondary9.29e-071.04e-07Monitor/moduleDiethylene glycolprimary9.23e-071.01e-07Japanese Electric GridMethyl chloridemodel/secondary8.94e-071.00e-07US electric grid2-Chloroacetophenonemodel/secondary8.35e-079.34e-08US electric gridBromomethanemodel/secondary8.27e-079.24e-08Japanese Electric GridToluenemodel/secondary8.04e-078.89e-08LPG ProductionSiliconsecondary8.04e-078.89e-08LPG ProductionCyanide (-1)secondary8.04e-077.84e-08Japanese Electric GridToluenemodel/secondary8.04e-077.84e-08Japanese Electric GridMethyl hydrazinemodel/secondary6.85e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary5.35e-085.39e-076.63e-08Natural Gas Prod.Animonysecondary5.36e-075.31e-075.31e-08LPG ProductionLeadsecondary5.31e-075.31e-08LPG Pr	Japanese Electric Grid	Acrolein	model/secondary	1.11e-06	1.25e-07
Monitor/moduleChromium (VI)primary1.09e-061.22e-07Fuel Oil #2 Prod.Scleniumsecondary1.09e-061.22e-07Fuel Oil #6 Prod.Hydrogen sulfidesecondary1.07e-061.20e-07US electric gridMolybdenummodel/secondary9.61e-071.07e-07Fuel Oil #4 Prod.Molybdenumsecondary9.29e-071.07e-07Fuel Oil #4 Prod.Molybdenumsecondary9.23e-071.03e-07Japanese Electric GridMethyl chloridemodel/secondary9.03e-071.01e-07JVS electric grid2-Chloroacetophenonemodel/secondary8.35e-079.34e-08US electric grid2-Chloroacetophenonemodel/secondary8.27e-079.24e-08LPG ProductionSiliconsecondary8.27e-079.24e-08LPG ProductionSiliconsecondary8.04e-078.99e-08Japanese Electric GridToluenemodel/secondary8.04e-078.99e-08LPG ProductionCyanide (-1)secondary8.04e-078.99e-08LPG ProductionBarium empdssecondary7.62e-078.52e-08Natural Gas Prod.Antimonysecondary6.58e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary5.39e-075.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary5.39e-075.59e-08Japanese Electric GridMethyl hydrazinemodel/secondary5.39e-075.57e-08Jeul Oil #6 Prod. <td< td=""><td>Fuel Oil #2 Prod.</td><td>Fluorides (F-)</td><td>secondary</td><td>1.09e-06</td><td>1.22e-07</td></td<>	Fuel Oil #2 Prod.	Fluorides (F-)	secondary	1.09e-06	1.22e-07
Fuel Oil #2 Prod.Seleniumsecondary1.09e-061.22e-07Fuel Oil #6 Prod.Hydrogen sulfidesecondary1.08e-061.22e-07LPG ProductionManganesesecondary1.07e-061.20e-07US clectrie gridMolybdenummodel/secondary9.61e-071.07e-07Fuel Oil #4 Prod.Molybdenumsecondary9.29e-071.04e-07Monitor/moduleDiethylene glycolprimary9.23e-071.04e-07Japanese Electric GridMethyl chloridemodel/secondary8.94e-071.00e-07US electric gridBariumsecondary8.35e-079.34e-08US electric gridBornomethanemodel/secondary8.37e-079.24e-08LPG ProductionSiliconsecondary8.12e-079.08e-08LPG ProductionCyanide (-1)secondary8.03e-078.99e-08LPG ProductionCyanide (-1)secondary8.03e-078.99e-08LPG ProductionCyanide (-1)secondary6.85e-077.66e-08Japanese Electric GridToluenemodel/secondary6.85e-077.66e-08LPG ProductionBarium empdssecondary5.08e-077.66e-08Japanese Electric GridAntimonysecondary5.30e-075.39e-08LPG ProductionLeadsecondary5.30e-075.93e-08LPG ProductionLeadsecondary5.30e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary5.30e-075.79e-08 <td>Monitor/module</td> <td>Chromium (VI)</td> <td>primary</td> <td>1.09e-06</td> <td>1.22e-07</td>	Monitor/module	Chromium (VI)	primary	1.09e-06	1.22e-07
Fuel Oil #6 Prod.Hydrogen sulfidesecondary1.08e-061.21e-07LPG ProductionManganesesecondary1.07e-061.20e-07US electric gridMolybdenummodel/secondary9.29e-071.04e-07Monitor/moduleDiethylene glycolprimary9.23e-071.04e-07Japanese Electric GridMethyl chloridemodel/secondary9.03e-071.04e-07US electric grid2-Chloroacetophenonemodel/secondary8.35e-079.34e-08US electric grid2-Chloroacetophenonemodel/secondary8.32e-079.24e-08LPG ProductionSiliconsecondary8.12e-079.24e-08LPG ProductionCyanide (-1)secondary8.04e-078.99e-08LPG ProductionCyanide (-1)secondary8.04e-078.99e-08LPG ProductionBarium empdssecondary7.01e-077.84e-08Natural Gas Prod.Antimorysecondary7.01e-077.84e-08LPG ProductionLeadsecondary5.36e-076.63e-08Fuel Oil #6 Prod.Ammoniasecondary5.36e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary6.35e-077.66e-08Juel Oil #6 Prod.Ammoniasecondary5.30e-075.61e-08Juel Oil #6 Prod.Molybdenumsecondary5.30e-075.93e-08Fuel Oil #6 Prod.Hydrofluoric acidsecondary5.30e-075.93e-08Fuel Oil #6 Prod.Hydrofluoric acidsecondary <td< td=""><td>Fuel Oil #2 Prod.</td><td>Selenium</td><td>secondary</td><td>1.09e-06</td><td>1.22e-07</td></td<>	Fuel Oil #2 Prod.	Selenium	secondary	1.09e-06	1.22e-07
LPG ProductionManganesesecondary1.07e-061.20e-07US electric gridMolybdenummodel/secondary9.61e-071.07e-07Fuel Oil #4 Prod.Molybdenumsecondary9.23e-071.04e-07Japanese Electric GridMethyl chlorideprimary9.23e-071.01e-07LPG ProductionBariumsecondary8.94e-071.00e-07US electric gridDeromenthanemodel/secondary8.35e-079.34e-08LPG ProductionBariumsecondary8.35e-079.24e-08LPG ProductionSiliconsecondary8.27e-079.24e-08LPG ProductionSiliconsecondary8.03e-078.98e-08LPG ProductionCyanide (-1)secondary8.03e-078.98e-08LPG ProductionBarium empdssecondary7.62e-077.84e-08Natural Gas Prod.Antimonysecondary6.85e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary6.54e-077.31e-08LPG ProductionLeadsecondary5.95e-076.10e-08Japanese Electric GridMethyl hydrazinesecondary5.95e-075.95e-08Fuel Oil #0 Prod.Leadsecondary5.95e-075.95e-08Fuel Oil #0 Prod.Molybdenumsecondary5.95e-075.95e-08Fuel Oil #0 Prod.Molybdenumsecondary5.95e-075.95e-08Fuel Oil #1 Prod.Molybdenumsecondary5.95e-075.95e-08Fuel Oil #2 Prod.	Fuel Oil #6 Prod.	Hydrogen sulfide	secondary	1.08e-06	1.21e-07
US electric gridMolybdenummodel/secondary9.61e-071.07e-07Fuel Oil #4 Prod.Molybdenumsecondary9.23e-071.04e-07Monitor/moduleDiethylene glycolprimary9.23e-071.03e-07Japanese Electric GridMethyl chloridemodel/secondary9.03e-071.00e-07LPG ProductionBariumsecondary8.35e-079.34e-08US electric grid2-Chloroacetophenonemodel/secondary8.27e-079.24e-08LPG ProductionSiliconsecondary8.12e-079.24e-08Japanese Electric GridToluenemodel/secondary8.04e-078.99e-08Japanese Electric GridToluenemodel/secondary8.03e-078.89e-08LPG ProductionCyanide (-1)secondary8.05e-077.66e-07Natural Gas Prod.Antimonysecondary6.85e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary6.54e-077.31e-08LPG ProductionLeadsecondary5.93e-076.63e-08Monitor/moduleNickelprimary5.45e-076.10e-08Fuel Oil #0 Prod.Molybdenumsecondary5.10e-075.97e-08Fuel Oil #0 Prod.Molybdenumsecondary5.10e-075.75e-08Fuel Oil #0 Prod.Molybdenumsecondary5.10e-075.75e-08Fuel Oil #0 Prod.Molybdenumsecondary5.10e-075.71e-08Fuel Oil #0 Prod.Molybdenumsecondary5.10e-075.71e	LPG Production	Manganese	secondary	1.07e-06	1.20e-07
Fuel Oil #4 Prod.Molybdenumsecondary9.29e-071.04e-07Monitor/moduleDiethylene glycolprimary9.23e-071.03e-07Japanese Electric GridMethyl chloridemodel/secondary9.03e-071.01e-07US electric grid2-Chloroacetophenonemodel/secondary8.94e-071.00e-07US electric grid2-Chloroacetophenonemodel/secondary8.27e-079.24e-08LPG ProductionSiliconsecondary8.12e-079.98e-08LPG ProductionCyanide (-1)secondary8.03e-078.98e-08LPG ProductionCyanide (-1)secondary8.03e-078.98e-08LPG ProductionBarium empdssecondary7.62e-078.52e-08Natural Gas Prod.Antimonysecondary6.85e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary6.54e-077.31e-08LPG ProductionLeadsecondary5.30e-075.61e-08Fuel Oil #6 Prod.Motybdenumsecondary5.17e-075.79e-08Fuel Oil #4 Prod.Hydrofluoric acidsecondary5.11e-075.79e-08Fuel Oil #4 Prod.Hydrofluoric acidsecondary5.11e-075.79e-08Fuel Oil #4 Prod.Hydrogen sulfidesecondary5.10e-075.51e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Fuel Oil #2 Prod.Ammoniasecondary5.02e-075.61e-08Fuel Oil #2 Prod.Hydrofluoric acid <td>US electric grid</td> <td>Molybdenum</td> <td>model/secondary</td> <td>9.61e-07</td> <td>1.07e-07</td>	US electric grid	Molybdenum	model/secondary	9.61e-07	1.07e-07
Monitor/moduleDiethylene glycolprimary9.23e-071.03e-07Japanese Electric GridMethyl chloridemodel/secondary9.03e-071.01e-07LPG ProductionBariumsecondary8.94e-071.00e-07US electric grid2-Chloroacetophenonemodel/secondary8.35e-079.24e-08US electric gridBromomethanemodel/secondary8.27e-079.24e-08LPG ProductionSiliconsecondary8.04e-078.99e-08Japanese Electric GridToluenemodel/secondary8.04e-078.99e-08LPG ProductionCyanide (-1)secondary8.04e-078.89e-08LPG ProductionBarium empdssecondary7.01e-077.84e-08LPG ProductionAntimonysecondary6.85e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary6.54e-077.31e-08LPG ProductionLeadsecondary5.30e-075.66-08Monitor/moduleNickelprimary5.45e-075.57e-08Fuel Oil #0 Frod.Molybdenumsecondary5.11e-075.79e-08Fuel Oil #1 Prod.Hydrofluoric acidsecondary5.11e-075.79e-08Fuel Oil #2 Prod.Arimoniasecondary5.02e-075.61e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridPropionaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary	Fuel Oil #4 Prod.	Molybdenum	secondary	9.29e-07	1.04e-07
Japanese Electric GridMethyl chloridemodel/secondary9.03e-071.01e-07LPG ProductionBariumsecondary8.94e-071.00e-07US electric grid2-Chloroacetophenonemodel/secondary8.35e-079.34e-08LPG ProductionSiliconsecondary8.12e-079.08e-08Japanese Electric GridToluenemodel/secondary8.04e-078.99e-08LPG ProductionCyanide (-1)secondary8.03e-078.98e-08LPG ProductionCyanide (-1)secondary7.01e-077.84e-08LPG ProductionBarium empdssecondary7.01e-077.84e-08Natural Gas Prod.Antimonysecondary6.85e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary6.54e-077.31e-08LPG ProductionLeadsecondary5.93e-076.63e-08LPG DroductionLeadsecondary5.30e-075.93e-08LPG IProductionLeadsecondary5.17e-075.79e-08Fuel Oil #0 Frod.Molybdenumsecondary5.17e-075.79e-08Fuel Oil #1 Prod.Hydrofluoric acidsecondary5.17e-075.79e-08Fuel Oil #2 Prod.Ammoniasecondary5.10e-075.71e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridAcetaldehydemodel/secondary5.02	Monitor/module	Diethylene glycol	primary	9.23e-07	1.03e-07
LPG ProductionBariumsecondary8.94e-071.00e-07US electric grid2-Chloroacetophenonemodel/secondary8.35e-079.34e-08US electric gridBromomethanemodel/secondary8.27e-079.24e-08LPG ProductionSiliconsecondary8.12e-079.08e-08Japanese Electric GridToluenemodel/secondary8.04e-078.98e-08LPG ProductionCyanide (-1)secondary8.04e-078.98e-08LPG ProductionBarium empdssecondary7.62e-078.52e-08Natural Gas Prod.Antimonysecondary6.85e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary6.54e-077.31e-08LPG ProductionLeadsecondary5.30e-075.93e-08LPG ProductionLeadsecondary5.30e-075.93e-08Fuel Oil #6 Prod.Molybdenumsecondary5.30e-075.93e-08Fuel Oil #2 Prod.Molybdenumsecondary5.17e-075.79e-08Fuel Oil #4 Prod.Hydrofluoria caidsecondary5.14e-075.75e-08Fuel Oil #2 Prod.Ammoniasecondary5.02e-075.61e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCamiummodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary5.	Japanese Electric Grid	Methyl chloride	model/secondary	9.03e-07	1.01e-07
US electric grid2-Chloroacetophenonemodel/secondary8.35e-079.34e-08US electric gridBromomethanemodel/secondary8.27e-079.24e-08LPG ProductionSiliconsecondary8.12e-079.08e-08Japanese Electric GridToluenemodel/secondary8.04e-078.99e-08LPG ProductionCyanide (-1)secondary8.03e-078.98e-08LPG ProductionBarium empdssecondary7.01e-077.84e-08Fuel Oil #6 Prod.Ammoniasecondary6.54e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary6.54e-077.61e-08LPG ProductionLeadsecondary5.93e-076.63a-08Monitor/moduleNickelprimary5.45e-075.93e-08Fuel Oil #6 Prod.Molybdenumsecondary5.17e-075.79e-08Fuel Oil #4 Prod.Hydrofluoria caidsecondary5.17e-075.79e-08Fuel Oil #2 Prod.Ammoniasecondary5.10e-075.71e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary4.58e-075.12e-08Vuel Oil #2 Prod.Hydrogen sulfide <td< td=""><td>LPG Production</td><td>Barium</td><td>secondary</td><td>8.94e-07</td><td>1.00e-07</td></td<>	LPG Production	Barium	secondary	8.94e-07	1.00e-07
US electric gridBromomethanemodel/secondary8.27e-079.24e-08LPG ProductionSiliconsecondary8.12e-079.08e-08Japanese Electric GridToluenemodel/secondary8.04e-078.99e-08LPG ProductionCyanide (-1)secondary8.03e-078.98e-08LPG ProductionBarium empdssecondary7.02e-078.52e-08Natural Gas Prod.Antimonysecondary7.01e-077.84e-08Fuel Oil #6 Prod.Ammoniasecondary6.54e-077.31e-08LPG ProductionLeadsecondary5.93e-076.63a-08Monitor/moduleNickelprimary5.45e-075.93e-07Fuel Oil #6 Prod.Molybdenumsecondary5.30e-075.93e-08Fuel Oil #4 Prod.Hydrofluoric acidsecondary5.17e-075.75e-08Fuel Oil #4 Prod.Zinc (elemental)secondary5.02e-075.71e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/seco	US electric grid	2-Chloroacetophenone	model/secondary	8.35e-07	9.34e-08
LPG ProductionSiliconsecondary8.12e-079.08e-08Japanese Electric GridToluenemodel/secondary8.04e-078.99e-08LPG ProductionCyanide (-1)secondary8.03e-078.98e-08LPG ProductionBarium empdssecondary7.01e-077.84e-08Natural Gas Prod.Antimonysecondary6.85e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary6.54e-077.31e-08LPG ProductionLeadsecondary5.93e-076.63e-08Monitor/moduleNickelprimary5.45e-076.10e-08Fuel Oil #6 Prod.Molybdenumsecondary5.17e-075.79e-08Fuel Oil #4 Prod.Hydrofluoric acidsecondary5.14e-075.75e-08Fuel Oil #4 Prod.Zinc (elemental)secondary5.10e-075.71e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary4.72e-075.28e-08Japanese Electric GridCadmiummodel/secondary4.72e-075.28e-08Japanese Electric GridCadmiummodel/secondary4.72e-075.28e-08Japanese Electric GridCadmiummodel/secondary4.72e-075.28e-08Japanese Electric GridDimethylbenzanthracenesecondary4.53e-075.07e-08Japanese Electric GridDimeth	US electric grid	Bromomethane	model/secondary	8.27e-07	9.24e-08
Japanese Electric GridToluenemodel/secondary8.04e-078.99e-08LPG ProductionCyanide (-1)secondary8.03e-078.98e-08LPG ProductionBarium empdssecondary7.62e-078.52e-08Natural Gas Prod.Antimonysecondary7.01e-077.84e-08Fuel Oil #6 Prod.Ammoniasecondary6.54e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary5.93e-076.63e-08Monitor/moduleNickelprimary5.45e-076.10e-08Fuel Oil #6 Prod.Molybdenumsecondary5.30e-075.93e-08Fuel Oil #4 Prod.Hydrofluoric acidsecondary5.17e-075.79e-08Fuel Oil #4 Prod.Zinc (elemental)secondary5.10e-075.71e-08Japanese Electric GridAmmoniasecondary5.10e-075.71e-08Japanese Electric GridAcetaldehydemodel/secondary5.10e-075.71e-08Japanese Electric GridPropionaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary4.72e-075.28e-08Natural Gas Prod.Dimethylbenzanthracenesecondary4.53e-075.07e-08US electric gridCyanide (-1)model/secondary4.35e-075.07e-08Japanese Electric GridDimethylbenzanthracenesecondary4.36e-074.48e-08Japanese Electric	LPG Production	Silicon	secondary	8.12e-07	9.08e-08
LPG ProductionCyanide (-1)secondary8.03e-078.98e-08LPG ProductionBarium empdssecondary7.62e-078.52e-08Natural Gas Prod.Antimonysecondary7.01e-077.84e-08Fuel Oil #6 Prod.Ammoniasecondary6.85e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary6.54e-077.31e-08LPG ProductionLeadsecondary5.93e-076.63e-08Monitor/moduleNickelprimary5.45e-076.10e-08Fuel Oil #6 Prod.Molybdenumsecondary5.30e-075.93e-08Fuel Oil #4 Prod.Hydrofluoric acidsecondary5.14e-075.79e-08Fuel Oil #2 Prod.Ammoniasecondary5.10e-075.71e-08Fuel Oil #2 Prod.Zinc (elemental)secondary5.02e-075.61e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridPropionaldchydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary4.72e-075.28e-08Monitor/moduleTinprimary4.58e-075.10e-08Japanese Electric GridDimethylbenzanthracenesecondary4.53e-075.07e-08US electric gridCyanide (-1)model/secondary4.34e-074.85e-08Japanese Electric GridMercurymodel/secondary4.34e-074.86e-08Japanese Electric GridDinethylbenzanthracenes	Japanese Electric Grid	Toluene	model/secondary	8.04e-07	8.99e-08
LPG ProductionBarium empdssecondary7.62e-078.52e-08Natural Gas Prod.Antimonysecondary7.01e-077.84e-08Fuel Oil #6 Prod.Ammoniasecondary6.85e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary6.54e-077.31e-08LPG ProductionLeadsecondary5.93e-076.63e-08Monitor/moduleNickelprimary5.45e-076.10e-08Fuel Oil #6 Prod.Molybdenumsecondary5.30e-075.93e-07Fuel Oil #2 Prod.Hydrofluoric acidsecondary5.17e-075.79e-08Fuel Oil #2 Prod.Ammoniasecondary5.10e-075.71e-08Fuel Oil #2 Prod.Zinc (elemental)secondary5.02e-075.61e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridPropionaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary4.72e-075.28e-08Monitor/moduleTinprimary4.58e-075.12e-08Vuel Oil #2 Prod.Dimethylbenzanthracenesecondary4.38e-075.12e-08Japanese Electric GridCyanide (-1)model/secondary4.58e-075.07e-08US electric gridCyanide (-1)model/secondary3.97e-074.48e-08Japanese Electric GridDiinethylbenzanthracene	LPG Production	Cyanide (-1)	secondary	8.03e-07	8.98e-08
Natural Gas Prod.Antimonysecondary7.01e-077.84e-08Fuel Oil #6 Prod.Ammoniasecondary6.85e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary6.54e-077.31e-08LPG ProductionLeadsecondary5.93e-076.63e-08Monitor/moduleNickelprimary5.45e-076.10e-08Fuel Oil #6 Prod.Molybdenumsecondary5.30e-075.93e-08Fuel Oil #4 Prod.Hydrofluoric acidsecondary5.17e-075.79e-08Fuel Oil #2 Prod.Ammoniasecondary5.14e-075.75e-08Fuel Oil #4 Prod.Zinc (elemental)secondary5.02e-075.61e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary4.74e-075.30e-08Fuel Oil #2 Prod.Hydrogen sulfidesecondary4.72e-075.28e-08Japanese Electric GridCadmiummodel/secondary4.58e-075.12e-08Japanese Electric GridDimethylbenzanthracenesecondary4.53e-075.07e-08Sutural Gas Prod.Dimethylbenzanthracenesecondary4.53e-075.07e-08Japanese Electric GridMercurymodel/secondary4.34e-074.85e-08Japanese Electric GridMercurymodel/secondary3.37e-074.44e-08Japanese Electric Grid <td< td=""><td>LPG Production</td><td>Barium empds</td><td>secondary</td><td>7.62e-07</td><td>8.52e-08</td></td<>	LPG Production	Barium empds	secondary	7.62e-07	8.52e-08
Fuel Oil #6 Prod.Ammoniasecondary6.85e-077.66e-08Japanese Electric GridMethyl hydrazinemodel/secondary6.54e-077.31e-08LPG ProductionLeadsecondary5.93e-076.63e-08Monitor/moduleNickelprimary5.45e-076.10e-08Fuel Oil #6 Prod.Molybdenumsecondary5.30e-075.93e-08Fuel Oil #4 Prod.Hydrofluoric acidsecondary5.17e-075.79e-08Fuel Oil #2 Prod.Ammoniasecondary5.14e-075.75e-08Fuel Oil #2 Prod.Zinc (elemental)secondary5.10e-075.71e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridPropionaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary4.74e-075.30e-08Fuel Oil #2 Prod.Hydrogen sulfidesecondary4.72e-075.28e-08Japanese Electric GridCadmiummodel/secondary4.53e-075.10e-08Japanese Electric GridDimethylbenzanthracenesecondary4.53e-075.10e-08Vul S electric gridCyanide (-1)model/secondary4.53e-075.02e-07Japanese Electric GridMercurymodel/secondary4.34e-074.85e-08Japanese Electric GridMercurysecondary4.53e-075.02e-08Natural Gas Prod.Dimethylbenzanthracenesecondary4.53e-075.02e-07Vul S electric g	Natural Gas Prod.	Antimony	secondary	7.01e-07	7.84e-08
Japanese Electric GridMethyl hydrazinemodel/secondary6.54e-077.31e-08LPG ProductionLeadsecondary5.93e-076.63e-08Monitor/moduleNickelprimary5.45e-076.10e-08Fuel Oil #6 Prod.Molybdenumsecondary5.30e-075.93e-08Fuel Oil #4 Prod.Hydrofluoric acidsecondary5.17e-075.79e-08Fuel Oil #2 Prod.Ammoniasecondary5.14e-075.75e-08Fuel Oil #4 Prod.Zinc (elemental)secondary5.02e-075.61e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridPropionaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary4.74e-075.30e-08Fuel Oil #2 Prod.Hydrogen sulfidesecondary4.74e-075.30e-08Japanese Electric GridCadmiummodel/secondary4.72e-075.28e-08Monitor/moduleTinprimary4.58e-075.12e-08Natural Gas Prod.Dimethylbenzanthracenesecondary4.34e-074.85e-08Japanese Electric GridMercurymodel/secondary3.97e-074.44e-08Japanese Electric GridDi(2-ethylhexyl)phthalatemodel/secondary3.86e-074.31e-08Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	Fuel Oil #6 Prod.	Ammonia	secondary	6.85e-07	7.66e-08
LPG ProductionLeadsecondary5.93e-076.63e-08Monitor/moduleNickelprimary5.45e-076.10e-08Fuel Oil #6 Prod.Molybdenumsecondary5.30e-075.93e-08Fuel Oil #4 Prod.Hydrofluoric acidsecondary5.17e-075.79e-08Fuel Oil #2 Prod.Ammoniasecondary5.14e-075.75e-08Fuel Oil #4 Prod.Zinc (elemental)secondary5.10e-075.71e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridPropionaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary4.74e-075.30e-08Fuel Oil #2 Prod.Hydrogen sulfidesecondary4.72e-075.28e-08Monitor/moduleTinprimary4.58e-075.12e-08Natural Gas Prod.Dimethylbenzanthracenesecondary4.53e-075.07e-08US electric gridCyanide (-1)model/secondary3.97e-074.44e-08Japanese Electric GridMercurymodel/secondary3.97e-074.44e-08Japanese Electric GridDi(2-ethylhexyl)phthalatemodel/secondary3.86e-074.31e-08Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	Japanese Electric Grid	Methyl hydrazine	model/secondary	6.54e-07	7.31e-08
Monitor/moduleNickelprimary5.45e-076.10e-08Fuel Oil #6 Prod.Molybdenumsecondary5.30e-075.93e-08Fuel Oil #4 Prod.Hydrofluoric acidsecondary5.17e-075.79e-08Fuel Oil #2 Prod.Ammoniasecondary5.14e-075.75e-08Fuel Oil #4 Prod.Zinc (elemental)secondary5.10e-075.71e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridPropionaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary4.74e-075.30e-08Fuel Oil #2 Prod.Hydrogen sulfidesecondary4.72e-075.28e-08Monitor/moduleTinprimary4.58e-075.12e-08Natural Gas Prod.Dimethylbenzanthracenesecondary4.53e-075.07e-08US electric gridCyanide (-1)model/secondary3.97e-074.44e-08Japanese Electric GridMercurymodel/secondary3.97e-074.44e-08Japanese Electric GridDi(2-ethylhexyl)phthalatemodel/secondary3.86e-074.31e-08Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	LPG Production	Lead	secondary	5.93e-07	6.63e-08
Fuel Oil #6 Prod.Molybdenumsecondary5.30e-075.93e-08Fuel Oil #4 Prod.Hydrofluoric acidsecondary5.17e-075.79e-08Fuel Oil #2 Prod.Ammoniasecondary5.14e-075.75e-08Fuel Oil #4 Prod.Zinc (elemental)secondary5.10e-075.71e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridPropionaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary4.74e-075.30e-08Fuel Oil #2 Prod.Hydrogen sulfidesecondary4.72e-075.28e-08Monitor/moduleTinprimary4.58e-075.12e-08Natural Gas Prod.Dimethylbenzanthracenesecondary4.34e-074.85e-08Japanese Electric GridMercurymodel/secondary3.97e-074.44e-08Japanese Electric GridDi(2-ethylhexyl)phthalatemodel/secondary3.86e-074.31e-08Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	Monitor/module	Nickel	primary	5.45e-07	6.10e-08
Fuel Oil #4 Prod.Hydrofluoric acidsecondary5.17e-075.79e-08Fuel Oil #2 Prod.Ammoniasecondary5.14e-075.75e-08Fuel Oil #4 Prod.Zinc (elemental)secondary5.10e-075.71e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridPropionaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary4.74e-075.30e-08Fuel Oil #2 Prod.Hydrogen sulfidesecondary4.72e-075.28e-08Monitor/moduleTinprimary4.58e-075.12e-08Natural Gas Prod.Dimethylbenzanthracenesecondary4.34e-074.85e-08US electric gridCyanide (-1)model/secondary3.97e-074.44e-08Japanese Electric GridDi(2-ethylhexyl)phthalatemodel/secondary3.86e-074.31e-08Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	Fuel Oil #6 Prod.	Molybdenum	secondary	5.30e-07	5.93e-08
Fuel Oil #2 Prod.Ammoniasecondary5.14e-075.75e-08Fuel Oil #4 Prod.Zinc (elemental)secondary5.10e-075.71e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridPropionaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary4.74e-075.30e-08Fuel Oil #2 Prod.Hydrogen sulfidesecondary4.72e-075.28e-08Monitor/moduleTinprimary4.58e-075.12e-08Natural Gas Prod.Dimethylbenzanthracenesecondary4.53e-075.07e-08US electric gridCyanide (-1)model/secondary3.97e-074.44e-08Japanese Electric GridMercurymodel/secondary3.97e-074.44e-08Japanese Electric GridDi(2-ethylhexyl)phthalatemodel/secondary3.86e-074.31e-08Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	Fuel Oil #4 Prod.	Hydrofluoric acid	secondary	5.17e-07	5.79e-08
Fuel Oil #4 Prod.Zinc (elemental)secondary5.10e-075.71e-08Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridPropionaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary4.74e-075.30e-08Fuel Oil #2 Prod.Hydrogen sulfidesecondary4.72e-075.28e-08Monitor/moduleTinprimary4.58e-075.12e-08Natural Gas Prod.Dimethylbenzanthracenesecondary4.53e-075.07e-08US electric gridCyanide (-1)model/secondary3.97e-074.48e-08Japanese Electric GridMercurymodel/secondary3.97e-074.44e-08Japanese Electric GridDi(2-ethylhexyl)phthalatemodel/secondary3.86e-074.31e-08Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	Fuel Oil #2 Prod.	Ammonia	secondary	5.14e-07	5.75e-08
Japanese Electric GridAcetaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridPropionaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary4.74e-075.30e-08Fuel Oil #2 Prod.Hydrogen sulfidesecondary4.72e-075.28e-08Monitor/moduleTinprimary4.58e-075.12e-08Natural Gas Prod.Dimethylbenzanthracenesecondary4.53e-075.07e-08US electric gridCyanide (-1)model/secondary4.34e-074.85e-08Japanese Electric GridMercurymodel/secondary3.97e-074.44e-08Japanese Electric GridDi(2-ethylhexyl)phthalatemodel/secondary3.86e-074.31e-08Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	Fuel Oil #4 Prod.	Zinc (elemental)	secondary	5.10e-07	5.71e-08
Japanese Electric GridPropionaldehydemodel/secondary5.02e-075.61e-08Japanese Electric GridCadmiummodel/secondary4.74e-075.30e-08Fuel Oil #2 Prod.Hydrogen sulfidesecondary4.72e-075.28e-08Monitor/moduleTinprimary4.58e-075.12e-08Natural Gas Prod.Dimethylbenzanthracenesecondary4.53e-075.07e-08US electric gridCyanide (-1)model/secondary4.34e-074.85e-08Japanese Electric GridMercurymodel/secondary3.97e-074.44e-08Japanese Electric GridDi(2-ethylhexyl)phthalatemodel/secondary3.86e-074.31e-08Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	Japanese Electric Grid	Acetaldehyde	model/secondary	5.02e-07	5.61e-08
Japanese Electric GridCadmiummodel/secondary4.74e-075.30e-08Fuel Oil #2 Prod.Hydrogen sulfidesecondary4.72e-075.28e-08Monitor/moduleTinprimary4.58e-075.12e-08Natural Gas Prod.Dimethylbenzanthracenesecondary4.53e-075.07e-08US electric gridCyanide (-1)model/secondary4.34e-074.85e-08Japanese Electric GridMercurymodel/secondary3.97e-074.44e-08Japanese Electric GridDi(2-ethylhexyl)phthalatemodel/secondary3.86e-074.31e-08Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	Japanese Electric Grid	Propionaldehyde	model/secondary	5.02e-07	5.61e-08
Fuel Oil #2 Prod.Hydrogen sulfidesecondary4.72e-075.28e-08Monitor/moduleTinprimary4.58e-075.12e-08Natural Gas Prod.Dimethylbenzanthracenesecondary4.53e-075.07e-08US electric gridCyanide (-1)model/secondary4.34e-074.85e-08Japanese Electric GridMercurymodel/secondary3.97e-074.44e-08Japanese Electric GridDi(2-ethylhexyl)phthalatemodel/secondary3.86e-074.31e-08Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	Japanese Electric Grid	Cadmium	model/secondary	4.74e-07	5.30e-08
Monitor/moduleTinprimary4.58e-075.12e-08Natural Gas Prod.Dimethylbenzanthracenesecondary4.53e-075.07e-08US electric gridCyanide (-1)model/secondary4.34e-074.85e-08Japanese Electric GridMercurymodel/secondary3.97e-074.44e-08Japanese Electric GridDi(2-ethylhexyl)phthalatemodel/secondary3.86e-074.31e-08Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	Fuel Oil #2 Prod.	Hydrogen sulfide	secondary	4.72e-07	5.28e-08
Natural Gas Prod.Dimethylbenzanthracenesecondary4.53e-075.07e-08US electric gridCyanide (-1)model/secondary4.34e-074.85e-08Japanese Electric GridMercurymodel/secondary3.97e-074.44e-08Japanese Electric GridDi(2-ethylhexyl)phthalatemodel/secondary3.86e-074.31e-08Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	Monitor/module	Tin	primary	4.58e-07	5.12e-08
US electric gridCyanide (-1)model/secondary4.34e-074.85e-08Japanese Electric GridMercurymodel/secondary3.97e-074.44e-08Japanese Electric GridDi(2-ethylhexyl)phthalatemodel/secondary3.86e-074.31e-08Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	Natural Gas Prod.	Dimethylbenzanthracene	secondary	4.53e-07	5.07e-08
Japanese Electric GridMercurymodel/secondary3.97e-074.44e-08Japanese Electric GridDi(2-ethylhexyl)phthalatemodel/secondary3.86e-074.31e-08Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	US electric grid	Cyanide (-1)	model/secondary	4.34e-07	4.85e-08
Japanese Electric GridDi(2-ethylhexyl)phthalatemodel/secondary3.86e-074.31e-08Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	Japanese Electric Grid	Mercury	model/secondary	3.97e-07	4.44e-08
Fuel Oil #4 Prod.Ethanesecondary3.36e-073.76e-08	Japanese Electric Grid	Di(2-ethylhexyl)phthalate	model/secondary	3.86e-07	4.31e-08
•	Fuel Oil #4 Prod.	Ethane	secondary	3.36e-07	3.76e-08

 Table M-40.
 LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
US electric grid	2,3,7,8-TCDD	model/secondary	3.28e-07	3.67e-08
US electric grid	Nickel	model/secondary	3.16e-07	3.53e-08
LPG Production	Carbon disulfide	secondary	2.87e-07	3.21e-08
Fuel Oil #4 Prod.	Pentane	secondary	2.82e-07	3.15e-08
Fuel Oil #6 Prod.	Hydrofluoric acid	secondary	2.76e-07	3.08e-08
US electric grid	Naphthalene	model/secondary	2.62e-07	2.92e-08
Japanese Electric Grid	Hexane	model/secondary	2.58e-07	2.88e-08
Fuel Oil #4 Prod.	Phenol	secondary	2.51e-07	2.81e-08
Fuel Oil #2 Prod.	Molybdenum	secondary	2.49e-07	2.78e-08
Monitor/module	Manganese	primary	2.29e-07	2.56e-08
US electric grid	Barium	model/secondary	2.25e-07	2.52e-08
LPG Production	Benzyl chloride	secondary	2.25e-07	2.51e-08
Fuel Oil #6 Prod.	Zinc (elemental)	secondary	2.25e-07	2.51e-08
Natural Gas Prod.	Naphthalene	secondary	2.05e-07	2.29e-08
Fuel Oil #4 Prod.	Hexane	secondary	1.95e-07	2.18e-08
Natural Gas Prod.	Nickel	secondary	1.94e-07	2.17e-08
LPG Production	Aluminum (elemental)	secondary	1.86e-07	2.08e-08
Japanese Electric Grid	Dimethyl sulfate	model/secondary	1.85e-07	2.06e-08
LCD glass mfg.	Nitrate	primary	1.83e-07	2.05e-08
Japanese Electric Grid	Isophorone	model/secondary	1.77e-07	1.98e-08
LPG Production	Chloroform	secondary	1.75e-07	1.95e-08
Fuel Oil #4 Prod.	Nickel	secondary	1.66e-07	1.86e-08
Fuel Oil #2 Prod.	Zinc (elemental)	secondary	1.65e-07	1.85e-08
US electric grid	Carbon disulfide	model/secondary	1.55e-07	1.73e-08
Fuel Oil #6 Prod.	Ethane	secondary	1.51e-07	1.69e-08
Natural Gas Prod.	Chromium (VI)	secondary	1.49e-07	1.67e-08
Fuel Oil #2 Prod.	Hydrofluoric acid	secondary	1.47e-07	1.64e-08
Japanese Electric Grid	Methyl ethyl ketone	model/secondary	1.43e-07	1.60e-08
Japanese Electric Grid	Tetrachloroethylene	model/secondary	1.41e-07	1.57e-08
Fuel Oil #4 Prod.	PM-10	secondary	1.34e-07	1.50e-08
Fuel Oil #6 Prod.	Pentane	secondary	1.27e-07	1.42e-08
Japanese Electric Grid	Methyl methacrylate	model/secondary	1.22e-07	1.36e-08
US electric grid	Benzyl chloride	model/secondary	1.22e-07	1.36e-08
LPG Production	Cobalt	secondary	1.21e-07	1.35e-08
Monitor/module	Cadmium	primary	1 14e-07	1 28e-08
Japanese Electric Grid	Bervllium	model/secondary	1 09e-07	1 22e-08
Fuel Oil #2 Prod	Ethane	secondary	1.08e-07	1 20e-08
Natural Gas Prod	Hydrogen sulfide	secondary	1.06e-07	1.19e-08
Jananese Electric Grid	1 2-Dichloroethane	model/secondary	1.00e-07	1.19e-08
Jananese Electric Grid	Bromoform	model/secondary	9.97e-08	1.14e-08
Monitor/module	Mercury	nrimary	9.69e-08	1.11e 08
Jananese Electric Grid	Dichloromethane	model/secondary	9.62e-08	1.08e-08
Fuel Oil #6 Prod	Nickel	secondary	9.56e-08	1.00e 08
US alastria grid	Chloroform	model/secondem/	9.500-08	1.0/0-08
Evel Oil #2 Prod	Phenol	secondary	9.450-08	1.000-08
I PG Production	Acrolein	secondary	9.576-08	1.030-08
Natural Gas Prod	Barium	secondary	9.510-00	1.040-08
Fuel Oil #4 Prod	Antimony	secondary	9.508-08	1.040-08
Fuel Oil #2 Prod	Pantano	secondary	9.150-08	1.020-08
ruci Oli #2 Prod.	r cinaile Manganaga	secondary	9.02e-08	0.97-00
Us electric grid	2 Chloresestarlanen	model/secondary	8.83e-08	9.8/e-09
Natural Gas Prod.	2-Unioroacetopnenone	secondary	8.80e-08	9.84e-09

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Fuel Oil #6 Prod.	Hexane	secondary	8.78e-08	9.82e-09
Natural Gas Prod.	Bromomethane	secondary	8.71e-08	9.74e-09
Natural Gas Prod.	Copper	secondary	8.32e-08	9.31e-09
Fuel Oil #6 Prod.	Phenol	secondary	8.24e-08	9.21e-09
Japanese Electric Grid	Chlorobenzene	model/secondary	8.05e-08	9.01e-09
Fuel Oil #6 Prod.	PM-10	secondary	7.94e-08	8.88e-09
LPG Production	Methyl chloride	secondary	7.54e-08	8.43e-09
Panel components	Methyl ethyl ketone	primary	6.97e-08	7.79e-09
LPG Production	Beryllium	secondary	6.74e-08	7.54e-09
US electric grid	Chromium (VI)	model/secondary	6.65e-08	7.44e-09
Japanese Electric Grid	2,4-Dinitrotoluene	model/secondary	6.39e-08	7.14e-09
Fuel Oil #2 Prod.	Hexane	secondary	6.25e-08	6.99e-09
Natural Gas Prod.	Manganese	secondary	6.23e-08	6.97e-09
LPG Production	Methyl hydrazine	secondary	5.46e-08	6.10e-09
US electric grid	Acrolein	model/secondary	5.04e-08	5.63e-09
Fuel Oil #4 Prod.	Aluminum (+3)	secondary	4.98e-08	5.57e-09
Natural Gas Prod.	Cyanide (-1)	secondary	4.57e-08	5.12e-09
Monitor/module	Phenol	primary	4.54e-08	5.08e-09
Japanese Electric Grid	Cumene hydroperoxide	model/secondary	4.51e-08	5.05e-09
Fuel Oil #2 Prod.	Nickel	secondary	4.40e-08	4.93e-09
US electric grid	Copper	model/secondary	4.33e-08	4.84e-09
LPG Production	Acetaldehyde	secondary	4.19e-08	4.69e-09
LPG Production	Propionaldehyde	secondary	4.19e-08	4.69e-09
LPG Production	Mercury	secondary	4.17e-08	4.66e-09
US electric grid	Methyl chloride	model/secondary	4.08e-08	4.56e-09
Fuel Oil #6 Prod.	Antimony	secondary	4.03e-08	4.51e-09
LPG Production	Cadmium	secondary	3.89e-08	4.35e-09
US electric grid	Fluoride	model/secondary	3.78e-08	4.23e-09
Japanese Electric Grid	Ethylbenzene	model/secondary	3.49e-08	3.91e-09
Fuel Oil #2 Prod.	PM-10	secondary	3.47e-08	3.88e-09
Natural Gas Prod.	Lead	secondary	3.42e-08	3.82e-09
LPG Production	Cadmium cmpds	secondary	3.33e-08	3.73e-09
LPG Production	Di(2-ethylhexyl)phthalate	secondary	3.22e-08	3.60e-09
LPG Production	Toluene	secondary	2.99e-08	3.34e-09
US electric grid	Methyl hydrazine	model/secondary	2.95e-08	3.30e-09
Fuel Oil #2 Prod.	Antimony	secondary	2.95e-08	3.30e-09
Fuel Oil #4 Prod.	Chromium (VI)	secondary	2.60e-08	2.91e-09
US electric grid	Lead	model/secondary	2.47e-08	2.76e-09
Natural Gas Prod.	Nitrate	secondary	2.37e-08	2.65e-09
US electric grid	Cobalt	model/secondary	2.35e-08	2.63e-09
Natural Gas Prod.	Phenol	secondary	2.30e-08	2.57e-09
US electric grid	Propionaldehyde	model/secondary	2.27e-08	2.53e-09
US electric grid	Acetaldehyde	model/secondary	2.27e-08	2.53e-09
Fuel Oil #4 Prod.	Nitrate	secondary	2.26e-08	2.53e-09
Monitor/module	Tetrachloroethylene	primary	1.95e-08	2.18e-09
Fuel Oil #4 Prod.	Copper	secondary	1.92e-08	2.15e-09
Fuel Oil #2 Prod.	Aluminum (+3)	secondary	1.85e-08	2.07e-09
Japanese Electric Grid	Phenanthrene	model/secondary	1.75e-08	1.96e-09
US electric grid	Di(2-ethylhexyl)phthalate	model/secondary	1.74e-08	1.95e-09

 Table M-40.
 LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
LPG Production	1,4-Dichlorobenzene	secondary	1.71e-08	1.92e-09
Fuel Oil #4 Prod.	2-Chloroacetophenone	secondary	1.66e-08	1.85e-09
Fuel Oil #4 Prod.	Bromomethane	secondary	1.64e-08	1.84e-09
Natural Gas Prod.	Carbon disulfide	secondary	1.63e-08	1.83e-09
Fuel Oil #6 Prod.	Aluminum (+3)	secondary	1.63e-08	1.82e-09
Japanese Electric Grid	Methyl tert-butyl ether	model/secondary	1.60e-08	1.79e-09
LPG Production	Dimethyl sulfate	secondary	1.54e-08	1.72e-09
Japanese Electric Grid	Chromium (III)	model/secondary	1.51e-08	1.68e-09
LPG Production	Isophorone	secondary	1.48e-08	1.65e-09
US electric grid	Mercury	model/secondary	1.45e-08	1.63e-09
Fuel Oil #6 Prod.	Chromium (VI)	secondary	1.39e-08	1.56e-09
Fuel Oil #4 Prod.	Dimethylbenzanthracene	secondary	1.38e-08	1.55e-09
Natural Gas Prod.	Toluene	secondary	1.36e-08	1.52e-09
Natural Gas Prod.	Benzyl chloride	secondary	1.28e-08	1.43e-09
Japanese Electric Grid	1,1,1-Trichloroethane	model/secondary	1.23e-08	1.37e-09
Japanese Electric Grid	Phenol	model/secondary	1.22e-08	1.36e-09
LPG Production	Methyl ethyl ketone	secondary	1.19e-08	1.33e-09
LPG Production	Tetrachloroethylene	secondary	1.17e-08	1.31e-09
Japanese Electric Grid	Styrene	model/secondary	1.17e-08	1.31e-09
US electric grid	Hexane	model/secondary	1.16e-08	1.30e-09
Fuel Oil #4 Prod.	Naphthalene	secondary	1.15e-08	1.28e-09
Fuel Oil #4 Prod.	Manganese	secondary	1.15e-08	1.28e-09
Japanese Electric Grid	Vinyl acetate	model/secondary	1.14e-08	1.27e-09
Monitor/module	Trichloroethylene	primary	1.14e-08	1.27e-09
Fuel Oil #6 Prod.	Copper	secondary	1.08e-08	1.21e-09
LCD glass mfg.	Chromium	primary	1.02e-08	1.14e-09
LPG Production	Methyl methacrylate	secondary	1.02e-08	1.14e-09
Fuel Oil #4 Prod.	Silicon	secondary	1.01e-08	1.13e-09
Natural Gas Prod.	Chloroform	secondary	9.96e-09	1.11e-09
Japanese Electric Grid	Xylene (mixed isomers)	model/secondary	9.46e-09	1.06e-09
US electric grid	Cadmium	model/secondary	9.33e-09	1.04e-09
LCD glass mfg.	Nickel	primary	9.05e-09	1.01e-09
Fuel Oil #6 Prod.	2-Chloroacetophenone	secondary	8.84e-09	9.89e-10
Fuel Oil #6 Prod.	Bromomethane	secondary	8.75e-09	9.79e-10
Fuel Oil #4 Prod.	Cyanide (-1)	secondary	8.62e-09	9.64e-10
LPG Production	1,2-Dichloroethane	secondary	8.49e-09	9.50e-10
US electric grid	Dimethyl sulfate	model/secondary	8.33e-09	9.32e-10
LPG Production	Bromoform	secondary	8.32e-09	9.31e-10
Fuel Oil #6 Prod.	Nitrate	secondary	8.26e-09	9.24e-10
LPG Production	Chromium (III)	secondary	8.24e-09	9.21e-10
US electric grid	Toluene	model/secondary	8.23e-09	9.20e-10
Fuel Oil #2 Prod.	Nitrate	secondary	8.06e-09	9.01e-10
LPG Production	Dichloromethane	secondary	8.04e-09	8.99e-10
US electric grid	Isophorone	model/secondary	7.99e-09	8.94e-10
Fuel Oil #4 Prod.	Barium	secondary	7.83e-09	8.76e-10
Natural Gas Prod.	PM-10	secondary	7.79e-09	8.71e-10
Fuel Oil #2 Prod.	Chromium (VI)	secondary	7.39e-09	8.26e-10
Natural Gas Prod.	o-xylene	secondary	7.25e-09	8.11e-10
LPG Production	Chlorobenzene	secondary	6.72e-09	7.52e-10
LPG Production	Aromatic hydrocarbons	secondary	6.72e-09	7.51e-10
US electric grid	Methyl ethyl ketone	model/secondary	6.45e-09	7.21e-10

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Fuel Oil #4 Prod.	Lead	secondary	6.38e-09	7.13e-10
US electric grid	Tetrachloroethylene	model/secondary	6.35e-09	7.10e-10
LPG Production	Copper (+1 & +2)	secondary	6.29e-09	7.03e-10
Fuel Oil #6 Prod.	Dimethylbenzanthracene	secondary	6.21e-09	6.95e-10
Fuel Oil #6 Prod.	Manganese	secondary	6.09e-09	6.81e-10
Fuel Oil #6 Prod.	Silicon	secondary	5.97e-09	6.67e-10
Fuel Oil #4 Prod.	Barium cmpds	secondary	5.58e-09	6.24e-10
US electric grid	Methyl methacrylate	model/secondary	5.51e-09	6.16e-10
LPG Production	2,4-Dinitrotoluene	secondary	5.35e-09	5.98e-10
Fuel Oil #6 Prod.	Naphthalene	secondary	5.32e-09	5.95e-10
Natural Gas Prod.	Acrolein	secondary	5.31e-09	5.93e-10
Fuel Oil #2 Prod.	Copper	secondary	5.22e-09	5.84e-10
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	5.07e-09	5.67e-10
Japanese Electric Grid	Acenaphthene	model/secondary	4.94e-09	5.52e-10
Fuel Oil #2 Prod.	2-Chloroacetophenone	secondary	4.71e-09	5.27e-10
Fuel Oil #2 Prod.	Bromomethane	secondary	4.66e-09	5.22e-10
Japanese Electric Grid	Ethylene dibromide	model/secondary	4.61e-09	5.16e-10
Fuel Oil #6 Prod.	Cvanide (-1)	secondary	4.60e-09	5.14e-10
US electric grid	1.2-Dichloroethane	model/secondary	4.59e-09	5.14e-10
Natural Gas Prod.	Aluminum (+3)	secondary	4.51e-09	5.05e-10
US electric grid	Bromoform	model/secondary	4 50e-09	5 04e-10
Fuel Oil #2 Prod	Dimethylbenzanthracene	secondary	4 43e-09	4 95e-10
Natural Gas Prod	Beryllium	secondary	4 41e-09	4 93e-10
US electric grid	Dichloromethane	model/secondary	4.35e-09	4.86e-10
Japanese Electric Grid	o-xvlene	model/secondary	4 31e-09	4 83e-10
Natural Gas Prod	Methyl chloride	secondary	4 30e-09	4 81e-10
Natural Gas Prod	Cobalt	secondary	4 16e-09	4 66e-10
US electric grid	Bervllium	model/secondary	3 69e-09	4 13e-10
US electric grid	Chlorobenzene	model/secondary	3.64e-09	4 07e-10
Fuel Oil #2 Prod	Nanhthalene	secondary	3.60e-09	4 03e-10
Fuel Oil #6 Prod	Lead	secondary	3.40e-09	3.81e-10
Fuel Oil #6 Prod	Barium	secondary	3 39e-09	3.80e-10
L PG Production	o-xylene	secondary	3.29e-09	3.68e-10
Fuel Oil #2 Prod	Manganese	secondary	3.25e-09	3.65e-10
Natural Gas Prod	Methyl hydrazine	secondary	3.11e-09	3.48e-10
Japanese Electric Grid	Ethyl Chloride	model/secondary	3 08e-09	3.45e-10
Fuel Oil #4 Prod	Carbon disulfide	secondary	3.08e-09	3.43e-10
Natural Gas Prod	Mercury	secondary	2.97e-09	3.32e-10
US electric grid	2 4-Dinitrotoluene	model/secondary	2.97e-09	3.32e-10
Japanese Electric Grid	Benzo[a]anthracene	model/secondary	2.69e-09	3.01e-10
I PG Production	Ethylbenzene	secondary	2.05e-09	2 98e-10
Fuel Oil #2 Prod	Silicon	secondary	2.07C-09	2.986-10
Fuel Oil #2 Prod	Barium	secondary	2.000 09 2.56e-09	2.910 10 2.87e-10
Fuel Oil #2 Prod	Cyanide (1)	secondary	2.500-09	2.376-10
Fuel Oil #4 Prod	Benzyl chloride	secondary	2.430-09	2.740-10
Natural Gas Prod	A cetaldebyde	secondary	2.410-09	2.700-10
Natural Gas Prod	Dronionaldebydo	secondary	2.596-09	2.67-10
Fuel Oil #4 Prod	Aluminum (elementel)	secondary	2.596-09	2.076-10
Natural Gas Brad	Cadmium	secondary	2.516-09	2.300-10
matural Gas Prod.	Caumum	secondary	2.21e-09	2.4/e-10

 Table M-40.
 LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Fuel Oil #2 Prod.	Barium cmpds	secondary	2.08e-09	2.32e-10
Fuel Oil #4 Prod.	Chloroform	secondary	1.88e-09	2.10e-10
Natural Gas Prod.	Di(2-ethylhexyl)phthalate	secondary	1.84e-09	2.05e-10
Fuel Oil #6 Prod.	Barium cmpds	secondary	1.83e-09	2.04e-10
Fuel Oil #2 Prod.	Lead	secondary	1.81e-09	2.02e-10
Japanese Electric Grid	Chrysene	model/secondary	1.80e-09	2.01e-10
Fuel Oil #6 Prod.	Carbon disulfide	secondary	1.64e-09	1.84e-10
Japanese Electric Grid	Acetophenone	model/secondary	1.62e-09	1.81e-10
Japanese Electric Grid	Biphenyl	model/secondary	1.56e-09	1.74e-10
Japanese Electric Grid	Indeno(1,2,3-cd)pyrene	model/secondary	1.51e-09	1.68e-10
Japanese Electric Grid	Benzo[g,h,i]perylene	model/secondary	1.45e-09	1.62e-10
US electric grid	Ethylbenzene	model/secondary	1.43e-09	1.60e-10
Fuel Oil #6 Prod.	Aluminum (elemental)	secondary	1.37e-09	1.53e-10
LPG Production	Methyl tert-butyl ether	secondary	1.34e-09	1.50e-10
Fuel Oil #4 Prod.	Cobalt	secondary	1.33e-09	1.49e-10
Japanese Electric Grid	Benzo[b,j,k]fluoranthene	model/secondary	1.31e-09	1.46e-10
Monitor/module	1,1,1-Trichloroethane	primary	1.30e-09	1.45e-10
Fuel Oil #6 Prod.	Benzyl chloride	secondary	1.29e-09	1.44e-10
Japanese Electric Grid	Acenaphthylene	model/secondary	1.11e-09	1.24e-10
LPG Production	Phenanthrene	secondary	1.09e-09	1.22e-10
Fuel Oil #6 Prod.	Chloroform	secondary	1.00e-09	1.12e-10
Fuel Oil #4 Prod.	Acrolein	secondary	1.00e-09	1.12e-10
Japanese Electric Grid	Dibenzo[a,h]anthracene	model/secondary	9.95e-10	1.11e-10
LPG Production	Styrene	secondary	9.76e-10	1.09e-10
LPG Production	Vinyl acetate	secondary	9.52e-10	1.07e-10
LPG Production	3-Methylcholanthrene	secondary	8.99e-10	1.01e-10
Natural Gas Prod.	Dimethyl sulfate	secondary	8.78e-10	9.82e-11
Fuel Oil #2 Prod.	Carbon disulfide	secondary	8.75e-10	9.79e-11
Natural Gas Prod.	Isophorone	secondary	8.42e-10	9.42e-11
Natural Gas Prod.	Zinc (+2)	secondary	8.26e-10	9.24e-11
Fuel Oil #4 Prod.	Methyl chloride	secondary	8.10e-10	9.06e-11
Japanese Electric Grid	2-Methylnaphthalene	model/secondary	8.00e-10	8.94e-11
Natural Gas Prod.	Chlorine	secondary	7.46e-10	8.34e-11
US electric grid	Methyl tert-butyl ether	model/secondary	7.23e-10	8.09e-11
Fuel Oil #6 Prod.	Cobalt	secondary	7.23e-10	8.08e-11
Fuel Oil #4 Prod.	Bervllium	secondary	7.19e-10	8.04e-11
Fuel Oil #2 Prod.	Benzyl chloride	secondary	6.86e-10	7.67e-11
Natural Gas Prod.	Methyl ethyl ketone	secondary	6.79e-10	7.60e-11
Japanese Electric Grid	Pvrene	model/secondary	6.73e-10	7.52e-11
Natural Gas Prod.	Tetrachloroethylene	secondary	6.69e-10	7.48e-11
Fuel Oil #2 Prod.	Aluminum (elemental)	secondary	5.96e-10	6.67e-11
Fuel Oil #4 Prod.	Methyl hydrazine	secondary	5.86e-10	6.56e-11
Japanese Electric Grid	Fluorene	model/secondary	5.86e-10	6.55e-11
Natural Gas Prod	Silicon	secondary	5 85e-10	6 55e-11
Natural Gas Prod.	Methyl methacrylate	secondary	5.81e-10	6.49e-11
LPG Production	Zinc (+2)	secondary	5.63e-10	6.30e-11
Japanese Electric Grid	Fluoranthene	model/secondary	5 59e-10	6.26e-11
US electric grid	Phenol	model/secondary	5.51e-10	6.16e-11
Fuel Oil #2 Prod.	Chloroform	secondary	5 33e-10	5 96e-11
Fuel Oil #6 Prod	Acrolein	secondary	5 33e-10	5.96e-11
US electric grid	Styrene	model/secondary	5.28e-10	5 90e-11
		y	5.200 10	2.200 11

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
US electric grid	Vinyl acetate	model/secondary	5.15e-10	5.76e-11
Natural Gas Prod.	Barium cmpds	secondary	5.06e-10	5.66e-11
US electric grid	Phenanthrene	model/secondary	4.85e-10	5.43e-11
Natural Gas Prod.	1,2-Dichloroethane	secondary	4.84e-10	5.41e-11
Natural Gas Prod.	Bromoform	secondary	4.74e-10	5.31e-11
Natural Gas Prod.	Dichloromethane	secondary	4.58e-10	5.12e-11
Fuel Oil #4 Prod.	Acetaldehyde	secondary	4.50e-10	5.03e-11
Fuel Oil #4 Prod.	Propionaldehyde	secondary	4.50e-10	5.03e-11
Fuel Oil #4 Prod.	Mercury	secondary	4.46e-10	4.98e-11
Fuel Oil #6 Prod.	Methyl chloride	secondary	4.32e-10	4.83e-11
US electric grid	Xylene (mixed isomers)	model/secondary	4.27e-10	4.78e-11
Fuel Oil #4 Prod.	Cadmium	secondary	4.12e-10	4.60e-11
LPG Production	Ethylene dibromide	secondary	3.85e-10	4.31e-11
Natural Gas Prod.	Chlorobenzene	secondary	3.83e-10	4.29e-11
US electric grid	Chromium (III)	model/secondary	3.82e-10	4.27e-11
Fuel Oil #6 Prod.	Beryllium	secondary	3.81e-10	4.26e-11
Fuel Oil #2 Prod.	Cobalt	secondary	3.71e-10	4.15e-11
LPG Production	1,1,1-Trichloroethane	secondary	3.63e-10	4.06e-11
Fuel Oil #4 Prod.	Di(2-ethylhexyl)phthalate	secondary	3.46e-10	3.87e-11
Fuel Oil #6 Prod.	Methyl hydrazine	secondary	3.12e-10	3.49e-11
Natural Gas Prod.	2.4-Dinitrotoluene	secondary	3.05e-10	3.41e-11
Fuel Oil #4 Prod.	Toluene	secondary	2.92e-10	3.27e-11
LPG Production	2-Methylnaphthalene	secondary	2.88e-10	3.22e-11
Fuel Oil #2 Prod.	Acrolein	secondary	2.84e-10	3.18e-11
Natural Gas Prod.	3-Methylcholanthrene	secondary	2.66e-10	2.98e-11
LPG Production	Ethyl Chloride	secondary	2.57e-10	2.88e-11
LPG Production	Chlorine	secondary	2.55e-10	2.86e-11
Natural Gas Prod.	Chromium (III)	secondary	2.54e-10	2.84e-11
Fuel Oil #4 Prod.	Cadmium cmpds	secondary	2.44e-10	2.73e-11
Fuel Oil #6 Prod.	Acetaldehvde	secondary	2.40e-10	2.68e-11
Fuel Oil #6 Prod	Propionaldehyde	secondary	2.40e-10	2.68e-11
Fuel Oil #6 Prod.	Mercury	secondary	2.36e-10	2.64e-11
Fuel Oil #2 Prod	Methyl chloride	secondary	2.30e-10	2.57e-11
LPG Production	Cumene	secondary	2.18e-10	2.43e-11
Fuel Oil #6 Prod	Cadmium	secondary	2.13e-10	2.42e-11
US electric grid	1 1 1-Trichloroethane	model/secondary	2.10e-10	2.35e-11
US electric grid	Ethylene dibromide	model/secondary	2.08e-10	2.33e-11
Fuel Oil #2 Prod	Beryllium	secondary	2.00e-10	2.55e 11
Fuel Oil #6 Prod	Di(2-ethylhexyl)phthalate	secondary	1 84e-10	2.250 11 2.06e-11
Fuel Oil #2 Prod	Methyl hydrazine	secondary	1.67e-10	1.86e-11
Fuel Oil #4 Prod	Dimethyl sulfate	secondary	1.67e-10	1.85e-11
Fuel Oil #4 Prod	Isophorone	secondary	1.59e-10	1.05 c 11
Fuel Oil #4 Prod	1 4-Dichlorobenzene	secondary	1.55e-10	1.77e-11
Natural Gas Prod	Ethylbenzene	secondary	1.52e-10	1.70e-11
Japanese Electric Grid	Benzo[a]pyrene	model/secondary	1.320-10 1.46e-10	1.700 11
Fuel Oil #6 Prod	Toluene	secondary	1.43e-10	1.60e-11
US electric grid	Ethyl Chloride	model/secondary	1 39e-10	1.56e-11
LPG Production	Acetophenone	secondary	1.35e-10	1.50e-11
Natural Gas Prod	Aluminum (elemental)	secondary	1 34e-10	1.50e-11
		sees nam y	1.5-10-10	1.200 11

 Table M-40.
 LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
LPG Production	Biphenyl	secondary	1.30e-10	1.45e-11
Fuel Oil #4 Prod.	Methyl ethyl ketone	secondary	1.28e-10	1.43e-11
Fuel Oil #2 Prod.	Acetaldehyde	secondary	1.28e-10	1.43e-11
Fuel Oil #2 Prod.	Propionaldehyde	secondary	1.28e-10	1.43e-11
Fuel Oil #2 Prod.	Mercury	secondary	1.27e-10	1.42e-11
Fuel Oil #4 Prod.	Tetrachloroethylene	secondary	1.26e-10	1.41e-11
Fuel Oil #2 Prod.	Cadmium	secondary	1.18e-10	1.32e-11
US electric grid	Cumene	model/secondary	1.18e-10	1.32e-11
Natural Gas Prod.	Phenanthrene	secondary	1.11e-10	1.24e-11
Fuel Oil #4 Prod.	Methyl methacrylate	secondary	1.09e-10	1.22e-11
LPG Production	Acenaphthylene	secondary	1.03e-10	1.15e-11
Fuel Oil #2 Prod.	Di(2-ethylhexyl)phthalate	secondary	9.83e-11	1.10e-11
Fuel Oil #4 Prod.	1.2-Dichloroethane	secondary	9.12e-11	1.02e-11
Fuel Oil #2 Prod.	Cadmium cmpds	secondary	9.09e-11	1.02e-11
Fuel Oil #4 Prod.	Bromoform	secondary	8.94e-11	1.00e-11
Fuel Oil #2 Prod.	Toluene	secondary	8.85e-11	9.90e-12
Fuel Oil #6 Prod.	Dimethyl sulfate	secondary	8.82e-11	9.87e-12
Fuel Oil #4 Prod.	Dichloromethane	secondary	8.63e-11	9.65e-12
LPG Production	Acenaphthene	secondary	8 56e-11	9 57e-12
Natural Gas Prod	2-Methylnaphthalene	secondary	8 53e-11	9 54e-12
Fuel Oil #6 Prod	Isonhorone	secondary	8 46e-11	9.46e-12
Jananese Electric Grid	5-Methyl chrysene	model/secondary	8 46e-11	9.46e-12
Fuel Oil #6 Prod	Cadmium cmpds	secondary	8.00e-11	8 94e-12
Natural Gas Prod	Methyl tert-butyl ether	secondary	7.62e-11	8.52e-12
US electric grid	Acetophenone	model/secondary	7.02e 11 7.33e-11	8 19e-12
Fuel Oil #4 Prod	Chlorobenzene	secondary	7.55e 11 7.22e-11	8.08e-12
US electric grid	Binhenyl	model/secondary	7.22e-11 7.03e-11	7.86e-12
Fuel Oil #6 Prod	1 4 Dichlorobenzene	secondary	6.05e-11	7.86-12
Fuel Oil #6 Prod	Nethyl ethyl ketone	secondary	6.82e.11	7.780-12
Fuel Oil #6 Prod	Tatrachlaraethylana	secondary	6 720 11	7.030-12
L DC Droduction	Niekel empde	secondary	0.726-11	7.516-12
LPG Production	Panzolalanthracena	secondary	5.000-11	6 702 12
Ero Flouetion	Mathyl mathaemlate	secondary	5.996-11	6.52a.12
Fuel Oil #4 Prod	2 4 Dinitrataluana	secondary	5.75 . 11	6.420.12
Fuel OII #4 Prod.	2,4-Dimitrotoluene	secondary	5.75-11	0.43e-12
LPG Production	L and amm da	secondary	5./5e-11	6.43e-12
LPG Production	Lead cmpds	secondary	5.00e-11	6.20e-12
Natural Gas Prod.	Styrene	secondary	5.566-11	6.22e-12
Natural Gas Prod.	Vinyl acetate	secondary	5.43e-11	6.0/e-12
Fuel Oil #2 Prod.	1,4-Dichlorobenzene	secondary	4.95e-11	5.54e-12
Fuel Oil #4 Prod.	Aromatic hydrocarbons	secondary	4.92e-11	5.50e-12
Fuel Oil #6 Prod.	1,2-Dichloroethane	secondary	4.86e-11	5.44e-12
Fuel Oil #6 Prod.	Bromotorm	secondary	4.7/e-11	5.33e-12
Fuel Oil #2 Prod.	Dimethyl sulfate	secondary	4.70e-11	5.26e-12
Fuel Oil #4 Prod.	Copper $(+1 \& +2)$	secondary	4.60e-11	5.15e-12
Fuel Oil #6 Prod.	Dichloromethane	secondary	4.60e-11	5.15e-12
Fuel Oil #2 Prod.	Isophorone	secondary	4.51e-11	5.04e-12
LPG Production	Indeno(1,2,3-cd)pyrene	secondary	4.46e-11	4.98e-12
Fuel Oil #4 Prod.	Chromium (III)	secondary	4.44e-11	4.96e-12
US electric grid	Acenaphthylene	model/secondary	4.37e-11	4.88e-12
Fuel Oil #6 Prod.	Chlorobenzene	secondary	3.85e-11	4.31e-12
US electric grid	Acenaphthene	model/secondary	3.74e-11	4.18e-12

Table M-40. LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Fuel Oil #2 Prod.	Methyl ethyl ketone	secondary	3.64e-11	4.07e-12
Fuel Oil #2 Prod.	Tetrachloroethylene	secondary	3.58e-11	4.00e-12
LPG Production	Benzo[b,j,k]fluoranthene	secondary	3.53e-11	3.95e-12
LPG Production	Mercury compounds	secondary	3.39e-11	3.79e-12
LPG Production	Benzo[a]pyrene	secondary	3.28e-11	3.67e-12
LPG Production	Fluorene	secondary	3.17e-11	3.54e-12
Fuel Oil #2 Prod.	Methyl methacrylate	secondary	3.11e-11	3.48e-12
Fuel Oil #4 Prod.	o-xylene	secondary	3.10e-11	3.47e-12
Fuel Oil #6 Prod.	2,4-Dinitrotoluene	secondary	3.06e-11	3.42e-12
Fuel Oil #4 Prod.	Ethylbenzene	secondary	2.86e-11	3.20e-12
LPG Production	Pyrene	secondary	2.74e-11	3.06e-12
LPG Production	Benzo[g,h,i]perylene	secondary	2.66e-11	2.98e-12
Fuel Oil #2 Prod.	1,2-Dichloroethane	secondary	2.59e-11	2.90e-12
LPG Production	Fluoranthene	secondary	2.59e-11	2.89e-12
Fuel Oil #2 Prod.	Bromoform	secondary	2.54e-11	2.84e-12
Fuel Oil #2 Prod.	Dichloromethane	secondary	2.45e-11	2.74e-12
LPG Production	HALON-1301	secondary	2.44e-11	2.72e-12
LPG Production	Benzo[b]fluoranthene	secondary	2.39e-11	2.68e-12
Fuel Oil #6 Prod.	Chromium (III)	secondary	2.37e-11	2.65e-12
Natural Gas Prod.	Cadmium cmpds	secondary	2.21e-11	2.47e-12
Natural Gas Prod	Ethylene dibromide	secondary	2.20e-11	2.46e-12
Natural Gas Prod.	1.1.1-Trichloroethane	secondary	2.07e-11	2.31e-12
US electric grid	Benzolb i klfluoranthene	model/secondary	2.07e-11	2.30e-12
Fuel Oil #2 Prod	Chlorobenzene	secondary	2.05e-11	2.30e-12
US electric grid	Chrysene	model/secondary	1 98e-11	2.230 12 2.21e-12
Fuel Oil #2 Prod	Aromatic hydrocarbons	secondary	1.93e-11	2.21e 12
Japanese Electric Grid	Anthracene	model/secondary	1.85e-11	2.05e 12 2.04e-12
US electric grid	Benzolalanthracene	model/secondary	1.80e-11	2.04e 12
Fuel Oil #2 Prod	Copper $(+1 \& +2)$	secondary	1.000 11 1.72e-11	1.92e-12
I PG Production	Dibenzo[a h]anthracene	secondary	1.72e 11	1.92e-12
End filled from	2 4 Dinitrotoluene	secondary	1.700-11	1.900-12
Fuel Oil #6 Prod	Aromatic hydrocarbons	secondary	1.61e-11	1.80e-12
US electric grid	Fluorono	model/secondamy	1.550.11	1.800-12
Fuel Oil #6 Prod	Ethylbonzono	socondom/	1.520.11	1.730-12
Fuel Oil #6 Prod	Eurypoenzene $(+1, 8, +2)$	secondary	1.520-11	1.706-12
Natural Gas Brad	Ethyl Chlorida	secondary	1.310-11	1.640.12
Fuel Oil #6 Prod		secondary	1.4/6-11	1.046-12
Fuel Oil #4 Prod	0-Ayiciic Mothyl tort bytyl other	secondary	1.400-11	1.610.12
Fuel OII #4 Prod.	Indens(1,2,2, ad)nymana	model/accordemy	1.446-11	1.01e-12
Evel Oil #2 Bred	Chromium (III)	model/secondary	1.200-11	1.436-12
Fuel OII #2 Floa.	Chironnum (III)	secondary	1.200-11	1.416-12
Natural Gas Prod.		secondary	1.24e-11	1.396-12
	Fluorantnene	model/secondary	1.24e-11	1.38e-12
Fuel Oil #4 Prod.	Phenanthrene	secondary	1.13e-11	1.26e-12
Natural Gas Prod.	Acenaphthylene	secondary	1.10e-11	1.23e-12
	Styrene	secondary	1.05e-11	1.1/e-12
Fuel Oil #4 Prod.	Vinyl acetate	secondary	1.02e-11	1.14e-12
US electric grid	Pyrene	model/secondary	1.02e-11	1.14e-12
Fuel Oil #2 Prod.	o-xylene	secondary	9.63e-12	1.08e-12
Natural Gas Prod.	Benzo[a]anthracene	secondary	8.84e-12	9.88e-13

 Table M-40. LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Chrysene	secondary	8.52e-12	9.52e-13
Fuel Oil #2 Prod.	Ethylbenzene	secondary	8.13e-12	9.09e-13
Fuel Oil #4 Prod.	3-Methylcholanthrene	secondary	8.12e-12	9.08e-13
Natural Gas Prod.	Indeno(1,2,3-cd)pyrene	secondary	7.77e-12	8.69e-13
Natural Gas Prod.	Acetophenone	secondary	7.72e-12	8.63e-13
Fuel Oil #6 Prod.	Methyl tert-butyl ether	secondary	7.66e-12	8.56e-13
Natural Gas Prod.	Biphenyl	secondary	7.40e-12	8.28e-13
US electric grid	o-xylene	model/secondary	7.35e-12	8.22e-13
Natural Gas Prod.	Acenaphthene	secondary	7.10e-12	7.94e-13
LPG Production	5-Methyl chrysene	secondary	7.06e-12	7.90e-13
US electric grid	Benzo[g,h,i]perylene	model/secondary	6.98e-12	7.81e-13
US electric grid	Benzo[a]pyrene	model/secondary	6.60e-12	7.38e-13
Natural Gas Prod.	Benzo[b]fluoranthene	secondary	6.58e-12	7.35e-13
Fuel Oil #6 Prod.	Phenanthrene	secondary	5.83e-12	6.52e-13
LPG Production	Halogenated matter (organic)	secondary	5.60e-12	6.26e-13
Fuel Oil #6 Prod.	Styrene	secondary	5.59e-12	6.25e-13
Fuel Oil #6 Prod.	Vinyl acetate	secondary	5.45e-12	6.10e-13
US electric grid	2-Methylnaphthalene	model/secondary	5.16e-12	5.78e-13
Natural Gas Prod.	Benzo[g,h,i]perylene	secondary	5.03e-12	5.63e-13
Natural Gas Prod.	Benzo[a]pyrene	secondary	4.96e-12	5.55e-13
Fuel Oil #4 Prod.	Zinc (+2)	secondary	4.60e-12	5.14e-13
Natural Gas Prod.	Dibenzo[a,h]anthracene	secondary	4.47e-12	5.00e-13
Natural Gas Prod.	Aromatic hydrocarbons	secondary	4.46e-12	4.99e-13
Natural Gas Prod.	Copper $(+1 \& +2)$	secondary	4.17e-12	4.67e-13
Fuel Oil #4 Prod.	Ethylene dibromide	secondary	4.14e-12	4.63e-13
Fuel Oil #2 Prod.	Methyl tert-butyl ether	secondary	4.08e-12	4.56e-13
Fuel Oil #4 Prod.	1.1.1-Trichloroethane	secondary	3.90e-12	4.36e-13
Natural Gas Prod.	Pyrene	secondary	3.86e-12	4.32e-13
US electric grid	5-Methyl chrysene	model/secondary	3.82e-12	4.27e-13
Fuel Oil #6 Prod.	3-Methylcholanthrene	secondary	3.65e-12	4.08e-13
Fuel Oil #2 Prod.	Phenanthrene	secondary	3.28e-12	3.67e-13
Fuel Oil #2 Prod.	Styrene	secondary	2.98e-12	3.33e-13
Fuel Oil #2 Prod.	Vinvl acetate	secondary	2.91e-12	3.25e-13
Fuel Oil #4 Prod.	Ethyl Chloride	secondary	2.76e-12	3.09e-13
Fuel Oil #4 Prod	2-Methylnaphthalene	secondary	2.60e-12	2.91e-13
Fuel Oil #2 Prod	3-Methylcholanthrene	secondary	2.60e-12	2.91e-13
Natural Gas Prod	Fluorene	secondary	2.58e-12	2.89e-13
Fuel Oil #4 Prod	Cumene	secondary	2.36e-12 2.34e-12	2.67e-13
Natural Gas Prod	Fluoranthene	secondary	2.34e-12 2.31e-12	2.52e-13
Fuel Oil #4 Prod	Chlorine	secondary	2.510 12 2.26e-12	2.53e-13
Fuel Oil #6 Prod	Ethylene dibromide	secondary	2.20e 12 2.21e-12	2.55e 13
Fuel Oil #6 Prod	1 1 1-Trichloroethane	secondary	2.210 12 2.08e-12	2.47e-13
Natural Gas Prod	Benzo[h i k]fluoranthene	secondary	2.00e 12 2.01e-12	2.52e-13
Fuel Oil #6 Prod	Zinc (+2)	secondary	1.81e-12	2.23e-13
US electric grid	Dibenzo[a h]anthracene	model/secondary	1.61e-12	1.89e-13
Eval Oil #2 Prod	Zinc (+2)	secondary	1.5% 12	1.77e 13
Fuel $Oil #6 Prod$	Ethyl Chloride	secondary	1.300-12	1.770-13
Fuel Oil #4 Prod	Acetophenone	secondary	1. 1 /c-12	1.63e-13
Fuel Oil #4 Prod	Binhenvl	secondary	1.400-12	1.030-13
Fuel Oil #6 Prod	Cumene	secondary	1.400-12	1.300-13
L DG Production	Anthracene	secondary	1.25e-12	1.396-13
	Anunatene	secondary	1.196-12	1.330-13

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Fuel Oil #2 Prod.	Ethylene dibromide	secondary	1.18e-12	1.31e-13
Fuel Oil #6 Prod.	2-Methylnaphthalene	secondary	1.17e-12	1.31e-13
Fuel Oil #2 Prod.	1,1,1-Trichloroethane	secondary	1.11e-12	1.24e-13
Fuel Oil #4 Prod.	Acenaphthylene	secondary	1.06e-12	1.19e-13
Fuel Oil #6 Prod.	Chlorine	secondary	1.01e-12	1.13e-13
Fuel Oil #2 Prod.	2-Methylnaphthalene	secondary	8.33e-13	9.31e-14
Fuel Oil #4 Prod.	Acenaphthene	secondary	8.30e-13	9.29e-14
Fuel Oil #2 Prod.	Ethyl Chloride	secondary	7.85e-13	8.78e-14
Fuel Oil #6 Prod.	Acetophenone	secondary	7.76e-13	8.67e-14
Fuel Oil #6 Prod.	Biphenyl	secondary	7.44e-13	8.32e-14
Fuel Oil #2 Prod.	Chlorine	secondary	7.28e-13	8.14e-14
Fuel Oil #2 Prod.	Cumene	secondary	6.64e-13	7.43e-14
Fuel Oil #4 Prod.	Chrysene	secondary	5.68e-13	6.35e-14
Fuel Oil #4 Prod.	Benzo[a]anthracene	secondary	5.64e-13	6.31e-14
Fuel Oil #6 Prod.	Acenaphthylene	secondary	5.49e-13	6.14e-14
Fuel Oil #4 Prod.	Nickel cmpds	secondary	4.88e-13	5.46e-14
US electric grid	Anthracene	model/secondary	4 49e-13	5.02e-14
Fuel Oil #4 Prod	Indeno(1.2.3-cd)nyrene	secondary	4 30e-13	4 81e-14
Fuel Oil #2 Prod	Acetophenone	secondary	4 13e-13	4.62e-14
Fuel Oil #4 Prod	Lead cmpds	secondary	4.10e-13	4.59e-14
Fuel Oil #6 Prod	A cenantthene	secondary	4.10e 13	4.51e-14
Natural Gas Prod	5-Methyl chrysene	secondary	4.03e-13	4.50e-14
Fuel Oil #2 Prod	Binhanyl	secondary	3.06e.13	4.300-14
Fuel Oil #4 Prod	Benzolh i klfluoranthene	secondary	3.70e-13	4.43C-14
Fuel Oil #4 Prod	Benzo[a]pyrene	secondary	3 380 13	3 780 14
Fuel Oil #4 Prod	Fluorono	secondary	2 220 12	3.780-14
Fuel Oil #2 Prod	A cenantithulana	secondary	3.100.13	3.720-14
Fuel Oil #6 Prod	Chrysone	secondary	2.810.12	3.140.14
Fuel Oil #4 Prod	Burene	secondary	2.010-13	3.146-14
Fuel Oil #4 Prod	Fluorenthene	secondary	2.740-13	3.010-14
Fuel Oil #6 Prod	Paraclalanthrasana	secondary	2.096-13	2.070.14
Fuel Oil #0 Prod.		secondary	2.03e-13	2.976-14
Fuel Oil #2 Prod.	Acenaphinene Deutse [] hill endeuts	secondary	2.556-13	2.836-14
Fuel Oil #4 Prod.	Menzolg, ii, i jperviene	secondary	2.496-13	2.796-14
Fuel Oil #4 Prod.	Mercury compounds	secondary	2.48e-13	2.78e-14
Fuel Oil #4 Prod.	Benzo[b]Iluoranthene	secondary	2.12e-13	2.37e-14
Fuel Oil #6 Prod.	Indeno(1,2,3-cd)pyrene	secondary	2.08e-13	2.33e-14
Fuel Oil #6 Prod.	Benzo[b,j,k]Iluorantnene	secondary	2.02e-13	2.26e-14
Japanese Electric Grid	2,3,7,8-TCDF	model/secondary	1.96e-13	2.20e-14
Fuel Oil #2 Prod.	Nickel cmpds	secondary	1.82e-13	2.03e-14
Fuel Oil #4 Prod.	HALON-1301	secondary	1.78e-13	2.00e-14
Fuel Oil #2 Prod.	Benzo[a]anthracene	secondary	1.75e-13	1.96e-14
Fuel Oil #6 Prod.	Fluorene	secondary	1.74e-13	1.94e-14
Fuel Oil #6 Prod.	Benzo[a]pyrene	secondary	1.74e-13	1.94e-14
Fuel Oil #2 Prod.	Chrysene	secondary	1.71e-13	1.91e-14
Fuel Oil #6 Prod.	Nickel empds	secondary	1.60e-13	1.79e-14
Fuel Oil #2 Prod.	Lead cmpds	secondary	1.53e-13	1.71e-14
Natural Gas Prod.	Anthracene	secondary	1.51e-13	1.69e-14
Fuel Oil #4 Prod.	Dibenzo[a,h]anthracene	secondary	1.49e-13	1.67e-14
Fuel Oil #6 Prod.	Fluoranthene	secondary	1.40e-13	1.56e-14

 Table M-40. LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Fuel Oil #6 Prod.	Pyrene	secondary	1.37e-13	1.54e-14
Fuel Oil #6 Prod.	Lead cmpds	secondary	1.34e-13	1.50e-14
Fuel Oil #2 Prod.	Indeno(1,2,3-cd)pyrene	secondary	1.31e-13	1.47e-14
Fuel Oil #6 Prod.	Benzo[g,h,i]perylene	secondary	1.17e-13	1.31e-14
Fuel Oil #2 Prod.	Benzo[b,j,k]fluoranthene	secondary	1.08e-13	1.21e-14
Fuel Oil #2 Prod.	Benzo[a]pyrene	secondary	9.87e-14	1.10e-14
Fuel Oil #2 Prod.	Fluorene	secondary	9.59e-14	1.07e-14
Fuel Oil #6 Prod.	Benzo[b]fluoranthene	secondary	9.33e-14	1.04e-14
Fuel Oil #2 Prod.	Mercury compounds	secondary	9.25e-14	1.04e-14
Fuel Oil #2 Prod.	Pyrene	secondary	8.17e-14	9.14e-15
Fuel Oil #6 Prod.	Mercury compounds	secondary	8.14e-14	9.10e-15
Fuel Oil #2 Prod.	Fluoranthene	secondary	7.81e-14	8.74e-15
Fuel Oil #2 Prod.	Benzo[g,h,i]perylene	secondary	7.78e-14	8.70e-15
Fuel Oil #4 Prod.	5-Methyl chrysene	secondary	7.59e-14	8.49e-15
Fuel Oil #2 Prod.	Benzo[b]fluoranthene	secondary	6.88e-14	7.70e-15
Fuel Oil #2 Prod.	HALON-1301	secondary	6.65e-14	7.44e-15
Fuel Oil #6 Prod.	Dibenzo[a,h]anthracene	secondary	6.48e-14	7.25e-15
Fuel Oil #6 Prod.	HALON-1301	secondary	5.85e-14	6.54e-15
Fuel Oil #2 Prod.	Dibenzo[a,h]anthracene	secondary	4.88e-14	5.46e-15
Natural Gas Prod.	Nickel cmpds	secondary	4.43e-14	4.95e-15
Fuel Oil #4 Prod.	Halogenated matter (organic)	secondary	4.10e-14	4.59e-15
Fuel Oil #6 Prod.	5-Methyl chrysene	secondary	4.04e-14	4.52e-15
Natural Gas Prod.	Lead cmpds	secondary	3.72e-14	4.16e-15
Natural Gas Prod.	Mercury compounds	secondary	2.25e-14	2.52e-15
Fuel Oil #2 Prod.	5-Methyl chrysene	secondary	2.16e-14	2.41e-15
Natural Gas Prod.	HALON-1301	secondary	1.62e-14	1.81e-15
Fuel Oil #2 Prod.	Halogenated matter (organic)	secondary	1.53e-14	1.71e-15
LPG Production	Halogenated hydrocarbons (unspecified)	secondary	1.40e-14	1.57e-15
Fuel Oil #6 Prod.	Halogenated matter (organic)	secondary	1.34e-14	1.50e-15
Fuel Oil #4 Prod.	Anthracene	secondary	1.21e-14	1.35e-15
US electric grid	2,3,7,8-TCDF	model/secondary	8.86e-15	9.90e-16
Fuel Oil #6 Prod.	Anthracene	secondary	6.10e-15	6.82e-16
Natural Gas Prod.	Halogenated matter (organic)	secondary	3.72e-15	4.16e-16
Fuel Oil #2 Prod.	Anthracene	secondary	3.56e-15	3.99e-16
Fuel Oil #4 Prod.	Halogenated hydrocarbons (unspecified)	secondary	1.02e-16	1.15e-17
Fuel Oil #2 Prod.	Halogenated hydrocarbons (unspecified)	secondary	3.82e-17	4.27e-18
Fuel Oil #6 Prod.	Halogenated hydrocarbons (unspecified)	secondary	3.36e-17	3.76e-18
Natural Gas Prod.	Halogenated hydrocarbons (unspecified)	secondary	9.28e-18	1.04e-18
Total Manufacturing			2.33e+02	2.60e+01
Use, Maintenance and Repair Life	e-cycle Stage			
US electric grid	Sulfur dioxide	model/secondary	6 15e+02	6 87e+01
US electric grid	Nitrogen oxides	model/secondary	4 39e-01	4 91e-02
US electric grid	Carbon monoxide	model/secondary	3 77e-01	4 22e-02
US electric grid	Methane	model/secondary	2 41e-01	2.69e-02
US electric grid	Arsenic	model/secondary	2.08e-01	2.09e 02
US electric grid	Hydrochloric acid	model/secondary	1.84e-01	2.05e-02
US electric grid	Selenium	model/secondary	3 46e-02	3 87e-03
US electric grid	Vanadium	model/secondary	2.61e-02	2 91e-03
US electric grid	PM-10	model/secondary	2.010-02 2.16e-02	2.910-03 2.41e-03
US electric grid	Hydrofluoric acid	model/secondary	5 022 03	5.620-04
ob electric griu	riyaronuone aciu	model/secondary	5.020-05	5.020-04

Table M-40. LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
US electric grid	Formaldehyde	model/secondary	3.62e-03	4.04e-04
US electric grid	Benzene	model/secondary	2.60e-03	2.91e-04
US electric grid	Phosphorus (yellow or white)	model/secondary	1.47e-03	1.64e-04
US electric grid	Nitrous oxide	model/secondary	1.27e-03	1.42e-04
US electric grid	Zinc (elemental)	model/secondary	6.77e-04	7.57e-05
US electric grid	Antimony	model/secondary	5.54e-04	6.19e-05
US electric grid	Molybdenum	model/secondary	1.85e-04	2.07e-05
US electric grid	2-Chloroacetophenone	model/secondary	1.61e-04	1.80e-05
US electric grid	Bromomethane	model/secondary	1.59e-04	1.78e-05
US electric grid	Cyanide (-1)	model/secondary	8.37e-05	9.36e-06
US electric grid	2,3,7,8-TCDD	model/secondary	6.33e-05	7.08e-06
US electric grid	Nickel	model/secondary	6.09e-05	6.81e-06
US electric grid	Naphthalene	model/secondary	5.04e-05	5.64e-06
US electric grid	Barium	model/secondary	4.35e-05	4.86e-06
US electric grid	Carbon disulfide	model/secondary	2.99e-05	3.34e-06
US electric grid	Benzyl chloride	model/secondary	2.34e-05	2.62e-06
US electric grid	Chloroform	model/secondary	1.82e-05	2.04e-06
US electric grid	Manganese	model/secondary	1 70e-05	1 90e-06
US electric grid	Chromium (VI)	model/secondary	1.78e-05	1.43e-06
US electric grid	Acrolein	model/secondary	9.71e-06	1.09e-06
US electric grid	Copper	model/secondary	8 34e-06	9.33e-07
US electric grid	Methyl chloride	model/secondary	7.86e-06	8 79e-07
US electric grid	Fluoride	model/secondary	7.000 00 7.29e-06	8.15e-07
US electric grid	Methyl hydrazine	model/secondary	5 69e-06	6.36e-07
US electric grid	Lead	model/secondary	4 76e-06	5.32e-07
US electric grid	Cobalt	model/secondary	4.700 00 4.54e-06	5.07e-07
US electric grid	Acetaldehyde	model/secondary	4.37e-06	4 89e-07
US electric grid	Propionaldehyde	model/secondary	4 37e-06	4 89e-07
US electric grid	Di(2-ethylhexyl)phthalate	model/secondary	3.36e-06	3.75e-07
US electric grid	Mercury	model/secondary	2.80e-06	3.14e-07
US electric grid	Heyane	model/secondary	2.000 00 2.24e-06	2.51e-07
US electric grid	Cadmium	model/secondary	1.80e-06	2.01e-07
US electric grid	Dimethyl sulfate	model/secondary	1.60e-06	1.80e-07
US electric grid	Toluene	model/secondary	1.59e-06	1.00 c 07
US electric grid	Isophorone	model/secondary	1.57e-06	1.776-07
US electric grid	Methyl ethyl ketone	model/secondary	1.34c-06	1.39e-07
US electric grid	Tetrachloroethylene	model/secondary	1.24e 00	1.37e-07
US electric grid	Methyl methacrylate	model/secondary	1.22e 00	1.19e-07
US electric grid	1 2-Dichloroethane	model/secondary	8 85e-07	9.90e-08
US electric grid	Bromoform	model/secondary	8.68e-07	9.71e-08
US electric grid	Dichloromethane	model/secondary	8.38e-07	9.376-08
US electric grid	Beryllium	model/secondary	7 12e-07	7.96e-08
US electric grid	Chlorobenzene	model/secondary	7.12e 07	7.90e 08
US electric grid	2 4-Dinitrotoluene	model/secondary	5 58e-07	6 24e-08
US electric grid	Ethylbenzene	model/secondary	2.76-07	3.000-08
US electric grid	Methyl tert-butyl ether	model/secondary	1 30- 07	1.56-08
US electric grid	Phenol	model/secondary	1.596-07	1.100.09
US electric grid	Styrene	model/secondary	1.000-07	1.190-00
US electric grid	Vinyl acetate	model/secondary	0.02-09	1.140-00
OS ciccuric griu	v myr acetate	model/secondary	9.958-08	1.116-08

 Table M-40.
 LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
US electric grid	Phenanthrene	model/secondary	9.35e-08	1.05e-08
US electric grid	Xylene (mixed isomers)	model/secondary	8.23e-08	9.21e-09
US electric grid	Chromium (III)	model/secondary	7.36e-08	8.23e-09
US electric grid	1,1,1-Trichloroethane	model/secondary	4.05e-08	4.53e-09
US electric grid	Ethylene dibromide	model/secondary	4.02e-08	4.49e-09
US electric grid	Ethyl Chloride	model/secondary	2.68e-08	3.00e-09
US electric grid	Cumene	model/secondary	2.27e-08	2.54e-09
US electric grid	Acetophenone	model/secondary	1.41e-08	1.58e-09
US electric grid	Biphenyl	model/secondary	1.35e-08	1.51e-09
US electric grid	Acenaphthylene	model/secondary	8.42e-09	9.41e-10
US electric grid	Acenaphthene	model/secondary	7.21e-09	8.06e-10
US electric grid	Benzo[b,j,k]fluoranthene	model/secondary	3.97e-09	4.44e-10
US electric grid	Chrysene	model/secondary	3.81e-09	4.26e-10
US electric grid	Benzo[a]anthracene	model/secondary	3.46e-09	3.87e-10
US electric grid	Fluorene	model/secondary	2.98e-09	3.34e-10
US electric grid	Indeno(1,2,3-cd)pyrene	model/secondary	2.46e-09	2.75e-10
US electric grid	Fluoranthene	model/secondary	2.38e-09	2.67e-10
US electric grid	Pyrene	model/secondary	1.97e-09	2.21e-10
US electric grid	o-xylene	model/secondary	1.42e-09	1.58e-10
US electric grid	Benzo[g,h,i]perylene	model/secondary	1.35e-09	1.50e-10
US electric grid	Benzo[a]pyrene	model/secondary	1.27e-09	1.42e-10
US electric grid	2-Methylnaphthalene	model/secondary	9.95e-10	1.11e-10
US electric grid	5-Methyl chrysene	model/secondary	7.36e-10	8.24e-11
US electric grid	Dibenzo[a,h]anthracene	model/secondary	3.26e-10	3.65e-11
US electric grid	Anthracene	model/secondary	8.65e-11	9.67e-12
US electric grid	2,3,7,8-TCDF	model/secondary	1.71e-12	1.91e-13
Total Use, Maintenance, and Repa	air		6.16e+02	6.89e+01
End-of-life Life-cycle Stage				
US electric grid	Sulfur dioxide	model/secondary	1.17e-01	1.30e-02
LCD landfilling	Sulfur dioxide	primary	4 45e-02	4 98e-03
LCD incineration	Sulfur dioxide	secondary	3.45e-02	3.86e-03
LCD landfilling	Carbon monoxide	primary	2.22e-03	2.48e-04
LCD landfilling	Nitrogen dioxide	primary	4.46e-04	4.99e-05
US electric grid	Nitrogen oxides	model/secondary	8.34e-05	9.32e-06
US electric grid	Carbon monoxide	model/secondary	7.15e-05	8.00e-06
LCD landfilling	РМ	primary	5.02e-05	5.61e-06
US electric grid	Methane	model/secondary	4.57e-05	5.11e-06
LCD incineration	Arsenic cmpds	secondary	4.30e-05	4.80e-06
US electric grid	Arsenic	model/secondary	3.95e-05	4.42e-06
LCD landfilling	Arsenic cmpds	primary	3.65e-05	4.08e-06
US electric grid	Hydrochloric acid	model/secondary	3.49e-05	3.90e-06
LPG Production	Carbon monoxide	secondary	2.11e-05	2.35e-06
LCD landfilling	Methane	primary	1.31e-05	1.47e-06
LCD landfilling	Benzene	primary	9.74e-06	1.09e-06
US electric grid	Selenium	model/secondary	6.57e-06	7.35e-07
US electric grid	Vanadium	model/secondary	4.94e-06	5.53e-07
LCD incineration	Lead	secondary	4.80e-06	5.37e-07
LPG Production	Vanadium	secondary	4.12e-06	4.61e-07
US electric grid	PM-10	model/secondarv	4.09e-06	4.58e-07
LPG Production	Benzene	secondary	3.38e-06	3.78e-07
		, J		

Table M-40. LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
LPG Production	Methane	secondary	3.30e-06	3.69e-07
LPG Production	Sulfur oxides	secondary	3.16e-06	3.53e-07
LCD landfilling	Silver compounds	primary	2.52e-06	2.82e-07
LCD incineration	Barium cmpds	secondary	2.33e-06	2.60e-07
LPG Production	Nitrogen oxides	secondary	2.26e-06	2.52e-07
LCD landfilling	Barium cmpds	primary	1.96e-06	2.19e-07
LCD incineration	Silver compounds	secondary	1.95e-06	2.19e-07
LCD landfilling	Hydrochloric acid	primary	1.13e-06	1.26e-07
US electric grid	Hydrofluoric acid	model/secondary	9.53e-07	1.07e-07
LPG Production	Arsenic	secondary	8.13e-07	9.10e-08
US electric grid	Formaldehyde	model/secondary	6.86e-07	7.67e-08
LCD landfilling	Ammonia	primary	5.79e-07	6.48e-08
LPG Production	Formaldehvde	secondary	5.28e-07	5.90e-08
LPG Production	PM	secondary	5.08e-07	5.68e-08
US electric grid	Benzene	model/secondary	4.94e-07	5.52e-08
LCD incineration	Cadmium cmpds	secondary	3 54e-07	3 96e-08
LCD landfilling	Cadmium empds	primary	3 27e-07	3.65e-08
US electric grid	Phosphorus (vellow or white)	model/secondary	2 78e-07	3 11e-08
US electric grid	Nitrous oxide	model/secondary	2.78e 07 2.41e-07	2.69e-08
LPG Production	Hydrochloric acid	secondary	1.45e-07	1.62e-08
US electric grid	Zinc (elemental)	model/secondary	1.45e 07	1.02e 00
US electric grid	Antimony	model/secondary	1.256-07	1.17e-08
LPG Production	Nitrous ovide	secondary	6 39e 08	7 15e 00
L CD landfilling	Hydrogen sulfide	primory	5 540 08	6 20e 09
LCD landming	Phosphorus (vellow or white)	secondary	3.860.08	4.320.09
LI G Hoddetion	Molybdonum	modal/sacondamy	2 510 08	2 02 0 00
US electric grid	2 Chloroacetophonono	model/secondary	3.510-08	3.936-09
US electric grid	2-Chloroacetophenone	model/secondary	3.030-08	3.426-09
L DC Braduation	Elverides (E.)	model/secondary	3.020-08	2.212.00
LPG Production	Fluorides (F-)	secondary	2.900-08	3.316-09
	Council (1)	secondary	2.926-08	3.276-09
US electric grid	Cyanide (-1)	model/secondary	1.596-08	1.786-09
		secondary	1.47e-08	1.05e-09
LPG Production	Hydrogen sulfide	secondary	1.21e-08	1.35e-09
US electric grid	2,3,7,8-TCDD	model/secondary	1.20e-08	1.34e-09
US electric grid	Nickel	model/secondary	1.15e-08	1.29e-09
LCD landfilling	Selenium	primary	1.05e-08	1.1/e-09
US electric grid	Naphthalene	model/secondary	9.56e-09	1.07e-09
US electric grid	Barium	model/secondary	8.25e-09	9.22e-10
LPG Production	Molybdenum	secondary	6.49e-09	7.25e-10
US electric grid	Carbon disulfide	model/secondary	5.67e-09	6.34e-10
LPG Production	Zinc (elemental)	secondary	4.72e-09	5.28e-10
LCD landfilling	Chromium (VI)	primary	4.65e-09	5.20e-10
LCD incineration	Mercury compounds	secondary	4.57e-09	5.11e-10
US electric grid	Benzyl chloride	model/secondary	4.45e-09	4.97e-10
LCD landfilling	Carbon tetrachloride	primary	4.04e-09	4.52e-10
LPG Production	Hydrofluoric acid	secondary	3.95e-09	4.42e-10
LCD landfilling	Mercury compounds	primary	3.95e-09	4.42e-10
US electric grid	Chloroform	model/secondary	3.46e-09	3.87e-10
US electric grid	Manganese	model/secondary	3.23e-09	3.61e-10

 Table M-40. LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
LCD incineration	Carbon tetrachloride	secondary	3.13e-09	3.50e-10
LCD landfilling	Chloroform	primary	3.13e-09	3.50e-10
LPG Production	Ethane	secondary	3.06e-09	3.42e-10
LPG Production	Phenol	secondary	2.82e-09	3.15e-10
LPG Production	Pentane	secondary	2.56e-09	2.87e-10
US electric grid	Chromium (VI)	model/secondary	2.43e-09	2.72e-10
LCD landfilling	Toluene	primary	1.98e-09	2.21e-10
US electric grid	Acrolein	model/secondary	1.84e-09	2.06e-10
LPG Production	Hexane	secondary	1.78e-09	1.99e-10
US electric grid	Copper	model/secondary	1.58e-09	1.77e-10
US electric grid	Methyl chloride	model/secondary	1.49e-09	1.67e-10
US electric grid	Fluoride	model/secondary	1.38e-09	1.55e-10
LPG Production	Nickel	secondary	1.14e-09	1.28e-10
US electric grid	Methyl hydrazine	model/secondary	1.08e-09	1.21e-10
US electric grid	Lead	model/secondary	9.03e-10	1.01e-10
LPG Production	PM-10	secondary	8.86e-10	9.91e-11
US electric grid	Cobalt	model/secondary	8.61e-10	9.63e-11
LPG Production	Antimony	secondary	8.42e-10	9.42e-11
US electric grid	Propionaldehyde	model/secondary	8.29e-10	9.27e-11
US electric grid	Acetaldehyde	model/secondary	8.29e-10	9.27e-11
US electric grid	Di(2-ethylhexyl)phthalate	model/secondary	6.37e-10	7.12e-11
LPG Production	Aluminum (+3)	secondary	5.58e-10	6.24e-11
US electric grid	Mercury	model/secondary	5.32e-10	5.95e-11
US electric grid	Hexane	model/secondary	4.26e-10	4.76e-11
LCD landfilling	Xylene (mixed isomers)	primary	4.08e-10	4.56e-11
LPG Production	Chromium (VI)	secondary	3.97e-10	4.44e-11
US electric grid	Cadmium	model/secondary	3.41e-10	3.82e-11
US electric grid	Dimethyl sulfate	model/secondary	3.05e-10	3.41e-11
US electric grid	Toluene	model/secondary	3.01e-10	3.36e-11
US electric grid	Isophorone	model/secondary	2.92e-10	3.27e-11
LCD landfilling	Tetrachloroethylene	primary	2.89e-10	3.23e-11
LCD incineration	Lead cmpds	secondary	2.59e-10	2.90e-11
LCD landfilling	Lead cmpds	primary	2.40e-10	2.69e-11
LPG Production	Nitrate	secondary	2.39e-10	2.67e-11
US electric grid	Methyl ethyl ketone	model/secondary	2.36e-10	2.67e-11
US electric grid	Tetrachloroethylene	model/secondary	2.30e-10	2.60e-11
LCD landfilling	1 2-Dichloroethane	nrimary	2.22e 10 2.24e-10	2.50 e -11
US electric grid	Methyl methacrylate	model/secondary	2.21e 10	2.31e 11
LCD landfilling	Trichloroethylene	primary	1.68e-10	1.88e-11
US electric grid	1.2-Dichloroethane	model/secondary	1.68e-10	1.88e-11
US electric grid	Bromoform	model/secondary	1.65e-10	1.80e 11
US electric grid	Dichloromethane	model/secondary	1.59e-10	1.78e-11
L PG Production	Copper	secondary	1.37e-10	1.780-11
US electric grid	Beryllium	model/secondary	1.370-10	1.546-11
US electric grid	Chlorobenzene	model/secondary	1.350-10	1.31e-11
L CD incincration	Trichloroothylono	secondem.	1.330-10	1.490-11
L DG Production	2 Chlorogestonhenone	secondary	1.510-10	1.400-11
LFG Production	2-Chloroacetophenone	secondary	1.2/e-10	1.42e-11
LFG Production	Dimethylbenzanthracene	secondary	1.200-10	1.41e-11
L CD landfilling		secondary	1.25e-10	1.40e-11
LCD landfilling	Einyidenzene	primary	1.0/e-10	1.19e-11
US electric grid	2,4-Dinitrotoluene	model/secondary	1.06e-10	1.18e-11

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
LPG Production	Naphthalene	secondary	1.02e-10	1.14e-11
LPG Production	Manganese	secondary	8.79e-11	9.83e-12
LPG Production	Barium	secondary	7.34e-11	8.21e-12
LPG Production	Silicon	secondary	6.66e-11	7.45e-12
LPG Production	Cyanide (-1)	secondary	6.59e-11	7.37e-12
LPG Production	Barium cmpds	secondary	6.25e-11	6.99e-12
US electric grid	Ethylbenzene	model/secondary	5.24e-11	5.86e-12
LPG Production	Lead	secondary	4.87e-11	5.44e-12
LCD landfilling	Dichloromethane	primary	4.68e-11	5.24e-12
US electric grid	Methyl tert-butyl ether	model/secondary	2.65e-11	2.96e-12
LPG Production	Carbon disulfide	secondary	2.35e-11	2.63e-12
US electric grid	Phenol	model/secondary	2.02e-11	2.25e-12
US electric grid	Styrene	model/secondary	1.93e-11	2.16e-12
US electric grid	Vinyl acetate	model/secondary	1.88e-11	2.11e-12
LPG Production	Benzyl chloride	secondary	1.84e-11	2.06e-12
US electric grid	Phenanthrene	model/secondary	1.77e-11	1.98e-12
US electric grid	Xylene (mixed isomers)	model/secondary	1.56e-11	1.75e-12
LPG Production	Aluminum (elemental)	secondary	1.52e-11	1.71e-12
LPG Production	Chloroform	secondary	1.43e-11	1.60e-12
US electric grid	Chromium (III)	model/secondary	1.40e-11	1.56e-12
LPG Production	Cobalt	secondary	9.89e-12	1.11e-12
LCD incineration	o-xylene	secondary	9 08e-12	1.02e-12
LCD landfilling	Chromium (III)	primary	7 92e-12	8 86e-13
US electric grid	1.1.1-Trichloroethane	model/secondary	7.68e-12	8.59e-13
LPG Production	Acrolein	secondary	7.64e-12	8.55e-13
US electric grid	Ethylene dibromide	model/secondary	7 62e-12	8 52e-13
LPG Production	Methyl chloride	secondary	6 19e-12	6 92e-13
LPG Production	Bervllium	secondary	5 53e-12	6 19e-13
US electric grid	Ethyl Chloride	model/secondary	5 09e-12	5 69e-13
LPG Production	Methyl hydrazine	secondary	4 48e-12	5.01e-13
US electric grid	Cumene	model/secondary	4 31e-12	4 82e-13
LPG Production	Acetaldehyde	secondary	3 44e-12	3 85e-13
LPG Production	Propionaldehyde	secondary	3 44e-12	3.85e-13
LPG Production	Mercury	secondary	3.42e-12	3.83e-13
LPG Production	Cadmium	secondary	3.12e-12	3 57e-13
LPG Production	Cadmium cmpds	secondary	2 74e-12	3.06e-13
US electric grid	Acetophenone	model/secondary	2.7 Te 12 2 68e-12	3.00e-13
LPG Production	Di(2-ethylbeyyl)phthalate	secondary	2.66e-12	2.96e-13
US electric grid	Binhenyl	model/secondary	2.04e 12 2.57e-12	2.900 13
LPG Production	Toluene	secondary	2.57e-12	2.87C-13
US electric grid	Acenanthylene	model/secondary	1.60e 12	1 70e 13
LPG Production	1 4-Dichlorobenzene	secondary	1.00c-12	1.57e-13
US electric grid	Acenanhthene	model/secondary	1.410-12	1.576-13
LPG Production	Dimethyl sulfate	secondary	1.376-12	1.11e 13
LPG Production	Isophorone	secondary	1.270-12	1.410-13
LPG Production	Methyl ethyl kotono	secondary	0.70-12	1.00-12
L PG Production	Tetrachloroethylono	secondary	9.790-13	1.090-13
L PG Production	Methyl methagylata	secondary	9.036-13	0.250.14
LI G FIOUUCION	Ponzo[h i k]flyorenthana	secondary	0.500-15	9.556-14
US electric grid	Benzolo,J,KJnuorantnene	model/secondary	/.55e-13	0.43e-14

 Table M-40.
 LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
US electric grid	Chrysene	model/secondary	7.23e-13	8.09e-14
LPG Production	1,2-Dichloroethane	secondary	6.97e-13	7.79e-14
LPG Production	Bromoform	secondary	6.83e-13	7.64e-14
LPG Production	Chromium (III)	secondary	6.76e-13	7.56e-14
LCD landfilling	Vinyl chloride	primary	6.68e-13	7.47e-14
LPG Production	Dichloromethane	secondary	6.60e-13	7.38e-14
US electric grid	Benzo[a]anthracene	model/secondary	6.57e-13	7.34e-14
US electric grid	Fluorene	model/secondary	5.66e-13	6.33e-14
LPG Production	Chlorobenzene	secondary	5.52e-13	6.17e-14
LPG Production	Aromatic hydrocarbons	secondary	5.52e-13	6.17e-14
LCD incineration	Vinyl chloride	secondary	5.18e-13	5.79e-14
LPG Production	Copper (+1 & +2)	secondary	5.16e-13	5.77e-14
US electric grid	Indeno(1,2,3-cd)pyrene	model/secondary	4.67e-13	5.22e-14
US electric grid	Fluoranthene	model/secondary	4.52e-13	5.06e-14
LPG Production	2,4-Dinitrotoluene	secondary	4.39e-13	4.91e-14
US electric grid	Pyrene	model/secondary	3.74e-13	4.18e-14
LPG Production	o-xylene	secondary	2.70e-13	3.02e-14
US electric grid	o-xylene	model/secondary	2.69e-13	3.00e-14
US electric grid	Benzo[g,h,i]pervlene	model/secondary	2.55e-13	2.85e-14
US electric grid	Benzofalpyrene	model/secondary	2.41e-13	2.70e-14
LPG Production	Ethylbenzene	secondary	2.19e-13	2.45e-14
US electric grid	2-Methylnaphthalene	model/secondary	1.89e-13	2.11e-14
US electric grid	5-Methyl chrysene	model/secondary	1 40e-13	1 56e-14
LPG Production	Methyl tert-butyl ether	secondary	1 10e-13	1 23e-14
LPG Production	Phenanthrene	secondary	8 93e-14	9.98e-15
LPG Production	Styrene	secondary	8 01e-14	8.96e-15
LPG Production	Vinyl acetate	secondary	7 82e-14	8 74e-15
LPG Production	3-Methylcholanthrene	secondary	7 38e-14	8 26e-15
US electric grid	Dibenzo[a h]anthracene	model/secondary	6 19e-14	6.92e-15
LPG Production	$Z_{inc}(+2)$	secondary	4 62e-14	5.17e-15
LPG Production	Ethylene dibromide	secondary	3 16e-14	3 54e-15
LPG Production	1 1 1-Trichloroethane	secondary	2.98e-14	3 33e-15
LPG Production	2-Methylnanhthalene	secondary	2.966 11 2.37e-14	2.65e-15
LPG Production	Ethyl Chloride	secondary	2.57e 14	2.05e-15
LPG Production	Chlorine	secondary	2.11e-14	2.30e-15
LPG Production	Cumene	secondary	1 79e-14	2.00e-15
US electric grid	Anthracene	model/secondary	1.79e 14	1.84e-15
I PG Production	Acetophenone	secondary	1.040-14	1.04c-15
LPG Production	Binhenvl	secondary	1.110-14	1.240-15
LPG Production	Accommonthyland	secondary	9.42o.15	0.430.16
LPC Production		secondary	7.022.15	9.43e-10
LPG Production	Niekal ampda	secondary	5.470.15	6 120 16
LPC Production	Received and the same	secondary	5.47C-15	0.12e-10
LPG Production	Characterie	secondary	4.920-13	5.30e-10
LPG Production	L and amenda	secondary	4.72e-15	5.2/e-16
LPC Production	Independent 2.2. adjuntations	secondary	4.000-15	1.00- 16
L DG Production	Panzolh i klfuorenthere	secondary	3.000-15	4.09e-16
LFG Production	Moreum compounds	secondary	2.906-15	3.24e-10
LPG Production	Nercury compounds	secondary	2.78e-15	5.11e-16
	Benzolajpyrene	secondary	2.69e-15	3.01e-16
LPG Production	Fluorene	secondary	2.60e-15	2.91e-16
LPG Production	Pyrene	secondary	2.25e-15	2.52e-16

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
LPG Production	Benzo[g,h,i]perylene	secondary	2.19e-15	2.44e-16
LPG Production	Fluoranthene	secondary	2.12e-15	2.37e-16
LPG Production	HALON-1301	secondary	2.00e-15	2.24e-16
LPG Production	Benzo[b]fluoranthene	secondary	1.96e-15	2.20e-16
LPG Production	Dibenzo[a,h]anthracene	secondary	1.40e-15	1.56e-16
LPG Production	5-Methyl chrysene	secondary	5.80e-16	6.48e-17
LPG Production	Halogenated matter (organic)	secondary	4.60e-16	5.14e-17
US electric grid	2,3,7,8-TCDF	model/secondary	3.24e-16	3.62e-17
LPG Production	Anthracene	secondary	9.78e-17	1.09e-17
LPG Production	Halogenated hydrocarbons (unspecified)	secondary	1.15e-18	1.29e-19
Natural Gas Prod.	Halogenated hydrocarbons (unspecified)	secondary	-1.95e-18	-2.18e-19
LCD incineration	Halogenated hydrocarbons (unspecified)	secondary	-2.06e-17	-2.31e-18
Fuel Oil #4 Prod.	Halogenated hydrocarbons (unspecified)	secondary	-4.72e-16	-5.28e-17
Natural Gas Prod.	Halogenated matter (organic)	secondary	-7.81e-16	-8.73e-17
Natural Gas Prod.	HALON-1301	secondary	-3.40e-15	-3.80e-16
Natural Gas Prod.	Mercury compounds	secondary	-4.73e-15	-5.29e-16
Natural Gas Prod.	Lead cmpds	secondary	-7.81e-15	-8.73e-16
Natural Gas Prod.	Nickel cmpds	secondary	-9.29e-15	-1.04e-15
LCD incineration	Halogenated matter (organic)	secondary	-1.19e-14	-1.33e-15
Natural Gas Prod.	Anthracene	secondary	-3.17e-14	-3.54e-15
LCD incineration	HALON-1301	secondary	-5.17e-14	-5.79e-15
Fuel Oil #4 Prod.	Anthracene	secondary	-5.55e-14	-6.21e-15
Natural Gas Prod	5-Methyl chrysene	secondary	-8 45e-14	-9 45e-15
LCD incineration	Nickel cmpds	secondary	-1.42e-13	-1.58e-14
Fuel Oil #4 Prod	Halogenated matter (organic)	secondary	-1 89e-13	-2 11e-14
LCD incineration	Dibenzola hlanthracene	secondary	-2.91e-13	-3 26e-14
Fuel Oil #4 Prod	5-Methyl chrysene	secondary	-3 50e-13	-3.91e-14
Natural Gas Prod	Benzo[h i k]fluoranthene	secondary	-4 23e-13	-4 73e-14
LCD incineration	Benzo[b]fluoranthene	secondary	-4 31e-13	-4 82e-14
Natural Gas Prod	Fluoranthene	secondary	-4 85e-13	-5 42e-14
Natural Gas Prod	Fluorene	secondary	-5 43e-13	-6.07e-14
Fuel Oil #4 Prod	Dibenzo[a h]anthracene	secondary	-6 88e-13	-7 70e-14
Natural Gas Prod	Pyrene	secondary	-8 10e-13	-9.06e-14
Fuel Oil #4 Prod	HALON-1301	secondary	-8 22e-13	-9 19e-14
Natural Gas Prod	Conner (+1 & +2)	secondary	-8 77e-13	-9 80e-14
Natural Gas Prod	Aromatic hydrocarbons	secondary	-0.77e-13	-1.05e-13
Natural Gas Prod	Dibenzo[a h]anthracene	secondary	-9.38e-13	-1.05e-13
Fuel Oil #4 Prod	Benzo[h]fluoranthene	secondary	0.780.13	1.09e 13
Natural Gas Prod	Benzo[a]nvrene	secondary	-)./8C-13	-1.096-13
Natural Gas Prod	Benzo[a h i]nervlene	secondary	-1.06e-12	-1 18e-13
Fuel Oil #4 Prod	Mercury compounds	secondary	-1.000-12	-1.180-13
Fuel Oil #4 Prod	Benzolg h ilpervlene	secondary	-1.140-12	-1.280-13
Fuel Oil #4 Prod	Fluoranthene	secondary	-1.130-12	-1.280-13
Fuel Oil #4 Prod	Purene	secondary	-1.240-12	-1.590-15
Natural Gas Prod	Pyrelie Benzo[b]fluoranthene	secondary	-1.200-12	-1.416-13
Natural Gas Prod		secondary	-1.500-12	-1.540-15
Eval Oil #4 Drod	Elucian	secondary	-1.496-12	-1.0/e-13
Fuct OII #4 FIOU.	Pinhonyl	secondary	-1.55-12	-1./1e-13
Tratural Gas Frou.	Diprienyi	secondary	-1.55e-12	-1./4e-13
ruei Oli #4 Prod.	Benzolajpyrene	secondary	-1.56e-12	-1./4e-13

 Table M-40. LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Acetophenone	secondary	-1.62e-12	-1.81e-13
Natural Gas Prod.	Indeno(1,2,3-cd)pyrene	secondary	-1.63e-12	-1.83e-13
Fuel Oil #4 Prod.	Benzo[b,j,k]fluoranthene	secondary	-1.75e-12	-1.96e-13
Natural Gas Prod.	Chrysene	secondary	-1.79e-12	-2.00e-13
Natural Gas Prod.	Benzo[a]anthracene	secondary	-1.86e-12	-2.08e-13
Fuel Oil #4 Prod.	Lead cmpds	secondary	-1.89e-12	-2.11e-13
Fuel Oil #4 Prod.	Indeno(1,2,3-cd)pyrene	secondary	-1.98e-12	-2.22e-13
Fuel Oil #4 Prod.	Nickel cmpds	secondary	-2.25e-12	-2.51e-13
Natural Gas Prod.	Acenaphthylene	secondary	-2.32e-12	-2.59e-13
LCD incineration	Anthracene	secondary	-2.41e-12	-2.70e-13
Fuel Oil #4 Prod.	Benzo[a]anthracene	secondary	-2.60e-12	-2.91e-13
Natural Gas Prod.	Cumene	secondary	-2.61e-12	-2.91e-13
Fuel Oil #4 Prod.	Chrysene	secondary	-2.61e-12	-2.92e-13
Natural Gas Prod.	Ethyl Chloride	secondary	-3.08e-12	-3.44e-13
Fuel Oil #4 Prod.	Acenaphthene	secondary	-3.83e-12	-4.28e-13
Natural Gas Prod.	1.1.1-Trichloroethane	secondary	-4.35e-12	-4.86e-13
Natural Gas Prod.	Ethylene dibromide	secondary	-4.61e-12	-5.16e-13
Natural Gas Prod.	Cadmium cmpds	secondary	-4.65e-12	-5.20e-13
Fuel Oil #4 Prod.	Acenaphthylene	secondary	-4.90e-12	-5.48e-13
LCD incineration	2-Methylnaphthalene	secondary	-5.65e-12	-6.31e-13
Fuel Oil #4 Prod.	Biphenyl	secondary	-6.43e-12	-7.19e-13
Fuel Oil #4 Prod.	Acetophenone	secondary	-6.71e-12	-7.50e-13
Fuel Oil #4 Prod	Chlorine	secondary	-1 04e-11	-1 17e-12
Fuel Oil #4 Prod	Cumene	secondary	-1 08e-11	-1 21e-12
Natural Gas Prod	Vinyl acetate	secondary	-1 14e-11	-1 27e-12
Natural Gas Prod	Styrene	secondary	-1 17e-11	-1 31e-12
Fuel Oil #4 Prod	2-Methylnaphthalene	secondary	-1 20e-11	-1 34e-12
LCD incineration	Chlorine	secondary	-1.26e-11	-1 40e-12
Fuel Oil #4 Prod	Ethyl Chloride	secondary	-1 27e-11	-1 42e-12
LCD incineration	Copper $(+1 \& +2)$	secondary	-1 34e-11	-1 49e-12
LCD incineration	Aromatic hydrocarbons	secondary	-1 43e-11	-1 60e-12
LCD incineration	Zinc (+2)	secondary	-1 45e-11	-1 62e-12
Natural Gas Prod	Methyl tert-butyl ether	secondary	-1.60e-11	-1 79e-12
I CD incineration	3-Methylcholanthrene	secondary	-1 76e-11	-1.97e-12
Natural Gas Prod	2-Methylnanhthalene	secondary	-1.70c-11	-1.97C-12
Fuel Oil #4 Prod	1.1.1-Trichloroethane	secondary	-1.7)c-11	-2.00c-12
Fuel Oil #4 Prod	Ethylene dibromide	secondary	-1.000-11	2 130 12
Fuel Oil #4 Prod	Zinc (+2)	secondary	-1.910-11	-2.130-12
LCD incineration	5 Methyl chrysene	secondary	-2.120-11	-2.37e-12
Network Cas Drod	Dhananthrana	secondary	-2.120-11	-2.376-12
Natural Gas Flod.	Prienanthrene Deutse [].	secondary	-2.55e-11	-2.01e-12
Natural Cas Brad	A huminum (alamantal)	secondary	-2.03e-11	-2.94e-12
Natural Gas Flod.	Dishlambarana (minadiaaman)	secondary	-2.010-11	-3.136-12
LCD incineration	Dichlorobenzene (mixed isomers)	secondary	-5.18e-11	-3.556-12
Natural Gas Prod.	Ethylbenzene	secondary	-3.20e-11	-3.58e-12
	Benzolajpyrene	secondary	-3.69e-11	-4.13e-12
	S-ivicing icholanthrene	secondary	-3./4e-11	-4.18e-12
Fuel Oil #4 Prod.	v inyl acetate	secondary	-4.71e-11	-5.27e-12
Fuel OII #4 Prod.	Styrene	secondary	-4.83e-11	-5.40e-12
	Pyrene	secondary	-5.06e-11	-5.66e-12
Fuel Oil #4 Prod.	Phenanthrene	secondary	-5.20e-11	-5.81e-12
Natural Gas Prod.	Chromium (III)	secondary	-5.33e-11	-5.96e-12

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Natural Gas Prod.	3-Methylcholanthrene	secondary	-5.59e-11	-6.25e-12
LCD incineration	Indeno(1,2,3-cd)pyrene	secondary	-5.92e-11	-6.62e-12
Natural Gas Prod.	2,4-Dinitrotoluene	secondary	-6.40e-11	-7.16e-12
LCD incineration	Fluoranthene	secondary	-6.52e-11	-7.29e-12
Fuel Oil #4 Prod.	Methyl tert-butyl ether	secondary	-6.62e-11	-7.40e-12
LCD incineration	Benzo[a]anthracene	secondary	-7.75e-11	-8.67e-12
Natural Gas Prod.	Chlorobenzene	secondary	-8.05e-11	-9.00e-12
LCD incineration	Fluorene	secondary	-8.35e-11	-9.34e-12
Natural Gas Prod.	Dichloromethane	secondary	-9.62e-11	-1.08e-11
LCD incineration	Chrysene	secondary	-9.68e-11	-1.08e-11
Natural Gas Prod.	Bromoform	secondary	-9.96e-11	-1.11e-11
Natural Gas Prod.	1,2-Dichloroethane	secondary	-1.02e-10	-1.14e-11
LCD incineration	Benzo[b,j,k]fluoranthene	secondary	-1.06e-10	-1.19e-11
Natural Gas Prod.	Barium cmpds	secondary	-1.06e-10	-1.19e-11
Natural Gas Prod.	Methyl methacrylate	secondary	-1.22e-10	-1.36e-11
Natural Gas Prod	Silicon	secondary	-1 23e-10	-1 37e-11
Fuel Oil #4 Prod	Ethylbenzene	secondary	-1 32e-10	-1 47e-11
Natural Gas Prod	Tetrachloroethylene	secondary	-1 40e-10	-1 57e-11
Natural Gas Prod	Methyl ethyl ketone	secondary	-1 43e-10	-1.60e-11
Fuel Oil #4 Prod	o-xylene	secondary	-1 43e-10	-1 60e-11
Natural Gas Prod	Chlorine	secondary	-1 57e-10	-1 75e-11
I CD incineration	Acenanhthene	secondary	-1.57e-10	-1.75c-11
Natural Gas Prod	Zing (+2)	secondary	-1.07e-10	1.8/e-11
Natural Gas Prod	Isophorope	secondary	-1.740-10	-1.940-11
Natural Gas Prod	Dimethyl sulfete	secondary	-1.770-10	2.062.11
Fuel Oil #4 Brod	Chromium (III)	secondary	-1.846-10	-2.00e-11
Fuel Oil #4 Prod.	Copport $(\pm 1, \pm 2)$	secondary	-2.046-10	2 270 11
Fuel Oil #4 Prod.	A normatic hydrogenhang	secondary	-2.126-10	-2.376-11
LCD incincration	A competitivitaria	secondary	-2.276-10	-2.346-11
Evol Oil #4 Prod	2.4 Dinitratelyana	secondary	-2.410-10	-2.706-11
Fuel Oil #4 Prod.	2,4-Dillitrototuene	secondary	-2.03e-10	-2.90e-11
Fuel Oli #4 Prod.	Di(2 attack area) white state	secondary	-3.536-10	-3.72e-11
Natural Gas Prod.	Di(2-ethyinexyi)phinalate	secondary	-3.85e-10	-4.31e-11
	Bipnenyi	secondary	-3.90e-10	-4.36e-11
Fuel OII #4 Prod.	Dichloromethane	secondary	-3.986-10	-4.45e-11
LCD incineration	Acetophenone	secondary	-4.0/e-10	-4.55e-11
Fuel Oil #4 Prod.	Bromotorm	secondary	-4.12e-10	-4.61e-11
Fuel Oil #4 Prod.	1,2-Dichloroethane	secondary	-4.20e-10	-4.70e-11
Natural Gas Prod.	Cadmium	secondary	-4.65e-10	-5.20e-11
Natural Gas Prod.	Propionaldehyde	secondary	-5.02e-10	-5.61e-11
Natural Gas Prod.	Acetaldehyde	secondary	-5.02e-10	-5.61e-11
Fuel Oil #4 Prod.	Methyl methacrylate	secondary	-5.04e-10	-5.64e-11
Fuel Oil #4 Prod.	Tetrachloroethylene	secondary	-5.81e-10	-6.50e-11
Fuel Oil #4 Prod.	Methyl ethyl ketone	secondary	-5.90e-10	-6.60e-11
Natural Gas Prod.	Mercury	secondary	-6.23e-10	-6.97e-11
Natural Gas Prod.	Methyl hydrazine	secondary	-6.53e-10	-7.31e-11
LCD incineration	Cumene	secondary	-6.53e-10	-7.31e-11
LCD incineration	Aluminum (elemental)	secondary	-7.02e-10	-7.85e-11
Fuel Oil #4 Prod.	1,4-Dichlorobenzene	secondary	-7.13e-10	-7.97e-11
Fuel Oil #4 Prod.	Isophorone	secondary	-7.31e-10	-8.18e-11

 Table M-40. LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Fuel Oil #4 Prod.	Dimethyl sulfate	secondary	-7.63e-10	-8.53e-11
LCD incineration	Ethyl Chloride	secondary	-7.72e-10	-8.64e-11
Natural Gas Prod.	Cobalt	secondary	-8.74e-10	-9.78e-11
Natural Gas Prod.	Methyl chloride	secondary	-9.03e-10	-1.01e-10
Natural Gas Prod.	Beryllium	secondary	-9.26e-10	-1.04e-10
Natural Gas Prod.	Aluminum (+3)	secondary	-9.48e-10	-1.06e-10
Natural Gas Prod.	1,4-Dichlorobenzene	secondary	-1.07e-09	-1.19e-10
LCD incineration	1,1,1-Trichloroethane	secondary	-1.09e-09	-1.22e-10
Natural Gas Prod.	Acrolein	secondary	-1.11e-09	-1.25e-10
Fuel Oil #4 Prod.	Cadmium cmpds	secondary	-1.12e-09	-1.26e-10
LCD incineration	Ethylene dibromide	secondary	-1.16e-09	-1.29e-10
Fuel Oil #4 Prod.	Toluene	secondary	-1.35e-09	-1.50e-10
Natural Gas Prod.	o-xylene	secondary	-1.52e-09	-1.70e-10
Fuel Oil #4 Prod.	Di(2-ethylhexyl)phthalate	secondary	-1.59e-09	-1.78e-10
Natural Gas Prod.	PM-10	secondary	-1.63e-09	-1.83e-10
Fuel Oil #4 Prod.	Cadmium	secondary	-1.90e-09	-2.12e-10
Fuel Oil #4 Prod.	Mercury	secondary	-2.05e-09	-2.30e-10
Fuel Oil #4 Prod.	Propionaldehyde	secondary	-2.07e-09	-2.32e-10
Fuel Oil #4 Prod.	Acetaldehyde	secondary	-2.07e-09	-2.32e-10
Natural Gas Prod.	Chloroform	secondary	-2.09e-09	-2.34e-10
LCD incineration	Xylene (mixed isomers)	secondary	-2.22e-09	-2.48e-10
LCD incineration	Phenanthrene	secondary	-2.61e-09	-2.91e-10
Natural Gas Prod.	Benzyl chloride	secondary	-2.69e-09	-3.01e-10
Fuel Oil #4 Prod.	Methyl hydrazine	secondary	-2.70e-09	-3.02e-10
Natural Gas Prod.	Toluene	secondary	-2.86e-09	-3.20e-10
LCD incineration	Vinyl acetate	secondary	-2.86e-09	-3.20e-10
LCD incineration	Styrene	secondary	-2.93e-09	-3.28e-10
LCD incineration	Silicon	secondary	-3.07e-09	-3.43e-10
Fuel Oil #4 Prod.	Beryllium	secondary	-3.31e-09	-3.71e-10
Natural Gas Prod.	Carbon disulfide	secondary	-3.43e-09	-3.84e-10
Fuel Oil #4 Prod.	Methyl chloride	secondary	-3.73e-09	-4.18e-10
LCD incineration	Methyl tert-butyl ether	secondary	-4.01e-09	-4.49e-10
Fuel Oil #4 Prod.	Acrolein	secondary	-4.61e-09	-5.15e-10
Natural Gas Prod.	Phenol	secondary	-4.83e-09	-5.40e-10
Natural Gas Prod.	Nitrate	secondary	-4.98e-09	-5.57e-10
Fuel Oil #4 Prod.	Cobalt	secondary	-6.12e-09	-6.85e-10
LCD incineration	Chloroacetophenone	secondary	-6.74e-09	-7.54e-10
Natural Gas Prod.	Lead	secondary	-7.17e-09	-8.02e-10
LCD incineration	Ethylbenzene	secondary	-7.87e-09	-8.80e-10
Fuel Oil #4 Prod	Chloroform	secondary	-8 65e-09	-9 67e-10
Natural Gas Prod	Cvanide (-1)	secondary	-9 61e-09	-1 07e-09
Fuel Oil #4 Prod	Aluminum (elemental)	secondary	-1.06e-08	-1 19e-09
Fuel Oil #4 Prod	Benzyl chloride	secondary	-1 11e-08	-1 24e-09
LCD incineration	Chromium (III)	secondary	-1 13e-08	-1 27e-09
Natural Gas Prod	Manganese	secondary	-1 31e-08	-1 46e-09
Fuel Oil #4 Prod	Carbon disulfide	secondary	-1 42e-08	-1 59e-09
LCD incineration	Aluminum (+3)	secondary	-1 44e-08	-1 61e-09
LCD incineration	2 4-Dinitrotoluene	secondary	-1.610.08	-1.80e-09
Natural Gas Prod	Conner	secondary	-1.750.08	-1.050-09
Natural Gas Prod	Bromomethane	secondary	-1.750-00	-2.05=.09
Natural Gas Prod	2-Chloroscetonhenone	secondary	-1.050-00	-2.030-09
ivatural Gas I fou.	2-Chloroacetophenone	secondary	-1.656-08	-2.076-09

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Barium	secondary	-1.95e-08	-2.18e-09
LCD incineration	Chlorobenzene	secondary	-2.02e-08	-2.26e-09
Natural Gas Prod.	Hydrogen sulfide	secondary	-2.23e-08	-2.49e-09
LCD incineration	Dichloromethane	secondary	-2.41e-08	-2.69e-09
LCD incineration	Bromoform	secondary	-2.50e-08	-2.79e-09
LCD incineration	1,2-Dichloroethane	secondary	-2.53e-08	-2.83e-09
Fuel Oil #4 Prod.	Barium cmpds	secondary	-2.57e-08	-2.87e-09
Fuel Oil #4 Prod.	Lead	secondary	-2.94e-08	-3.29e-09
LCD incineration	Methyl methacrylate	secondary	-3.06e-08	-3.42e-09
LCD incineration	Dimethylbenzanthracene	secondary	-3.12e-08	-3.49e-09
Natural Gas Prod.	Chromium (VI)	secondary	-3.13e-08	-3.50e-09
LCD incineration	Tetrachloroethylene	secondary	-3.50e-08	-3.91e-09
LCD incineration	Methyl ethyl ketone	secondary	-3.58e-08	-4.00e-09
Fuel Oil #4 Prod.	Barium	secondary	-3.61e-08	-4.04e-09
Fuel Oil #4 Prod.	Cyanide (-1)	secondary	-3.97e-08	-4.44e-09
Natural Gas Prod.	Nickel	secondary	-4.07e-08	-4.55e-09
LCD incineration	PM-10	secondary	-4.08e-08	-4.56e-09
Natural Gas Prod.	Naphthalene	secondary	-4.31e-08	-4.82e-09
LCD incineration	Isophorone	secondary	-4.43e-08	-4.96e-09
LCD incineration	Dimethyl sulfate	secondary	-4.62e-08	-5.17e-09
Fuel Oil #4 Prod.	Silicon	secondary	-4.64e-08	-5.19e-09
LCD incineration	Toluene	secondary	-4.73e-08	-5.29e-09
Fuel Oil #4 Prod.	Manganese	secondary	-5.28e-08	-5.91e-09
Fuel Oil #4 Prod.	Naphthalene	secondary	-5.29e-08	-5.91e-09
Fuel Oil #4 Prod.	Dimethylbenzanthracene	secondary	-6.37e-08	-7.12e-09
LCD incineration	Cadmium	secondary	-6.38e-08	-7.13e-09
Fuel Oil #4 Prod.	Bromomethane	secondary	-7.56e-08	-8.46e-09
LCD incineration	Phenol	secondary	-7.60e-08	-8.50e-09
Fuel Oil #4 Prod.	2-Chloroacetophenone	secondary	-7.64e-08	-8.55e-09
LCD incineration	Mercury	secondary	-8.42e-08	-9.42e-09
Fuel Oil #4 Prod.	Copper	secondary	-8.86e-08	-9.90e-09
Natural Gas Prod.	Dimethylbenzanthracene	secondary	-9.51e-08	-1.06e-08
LCD incineration	Di(2-ethylhexyl)phthalate	secondary	-9.66e-08	-1.08e-08
Fuel Oil #4 Prod.	Nitrate	secondary	-1.04e-07	-1.17e-08
Fuel Oil #4 Prod.	Chromium (VI)	secondary	-1.20e-07	-1.34e-08
LCD incineration	Cobalt	secondary	-1.24e-07	-1.39e-08
LCD incineration	Acetaldehvde	secondary	-1.26e-07	-1.41e-08
LCD incineration	Propionaldehyde	secondary	-1.26e-07	-1.41e-08
Natural Gas Prod.	Antimony	secondary	-1.47e-07	-1.65e-08
LCD incineration	Methyl hydrazine	secondary	-1.64e-07	-1.83e-08
LCD incineration	Beryllium	secondary	-1 94e-07	-2.18e-08
LCD incineration	Methyl chloride	secondary	-2.26e-07	-2.53e-08
Fuel Oil #4 Prod.	Aluminum (+3)	secondary	-2.29e-07	-2.56e-08
LCD incineration	Copper	secondary	-2.67e-07	-2.99e-08
LCD incineration	Acrolein	secondary	-2.79e-07	-3 12e-08
Fuel Oil #4 Prod	Antimony	secondary	-4 21e-07	-4 70e-08
LCD incineration	Hexane	secondary	-4 88e-07	-5 46e-08
LCD incineration	Hydrogen sulfide	secondary	-5 13e-07	-5 74e-08
LCD incineration	Chloroform	secondary	-5 22e-07	-5 84e-08
LCD memeration	Chiorotothi	secondary	5.220-07	5.040 00

 Table M-40. LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Natural Gas Prod.	Hydrofluoric acid	secondary	-5.77e-07	-6.45e-08
Natural Gas Prod.	Phosphorus (yellow or white)	secondary	-5.86e-07	-6.55e-08
LCD incineration	Pentane	secondary	-6.12e-07	-6.84e-08
Fuel Oil #4 Prod.	PM-10	secondary	-6.18e-07	-6.91e-08
LCD incineration	Benzyl chloride	secondary	-6.74e-07	-7.54e-08
LCD incineration	Ethane	secondary	-7.29e-07	-8.16e-08
Fuel Oil #4 Prod.	Nickel	secondary	-7.65e-07	-8.55e-08
Natural Gas Prod.	Molybdenum	secondary	-7.65e-07	-8.56e-08
LCD incineration	Carbon disulfide	secondary	-8.60e-07	-9.62e-08
LCD incineration	Barium	secondary	-8.87e-07	-9.92e-08
Fuel Oil #4 Prod.	Hexane	secondary	-9.00e-07	-1.01e-07
Fuel Oil #4 Prod.	Phenol	secondary	-1.16e-06	-1.30e-07
LCD incineration	Nitrate	secondary	-1.17e-06	-1.30e-07
Fuel Oil #4 Prod.	Pentane	secondary	-1.30e-06	-1.45e-07
LCD incineration	Naphthalene	secondary	-1.31e-06	-1.47e-07
Natural Gas Prod.	Hexane	secondary	-1.34e-06	-1.50e-07
Fuel Oil #4 Prod.	Ethane	secondary	-1.55e-06	-1.73e-07
Natural Gas Prod	Pentane	secondary	-1 94e-06	-2.17e-07
Natural Gas Prod	Ethane	secondary	-2.31e-06	-2.59e-07
Fuel Oil #4 Prod	Zinc (elemental)	secondary	-2.35e-06	-2.63e-07
Fuel Oil #4 Prod	Hydrofluoric acid	secondary	-2 38e-06	-2 67e-07
I CD incineration	Cvanide (-1)	secondary	-2.41e-06	-2 69e-07
Natural Gas Prod	Zinc (elemental)	secondary	-2.81e-06	-3 15e-07
I CD incineration	Manganese	secondary	-3 17e-06	-3 55e-07
Natural Gas Prod	Nitrous oxide	secondary	-3.82e-06	-4 28e-07
Natural Gas Prod	Selenium	secondary	-4 02e-06	-4 49e-07
Fuel Oil #4 Prod	Molyhdenum	secondary	-4.020-00	-4.79e-07
Natural Gas Prod	Fluorides (F)	secondary	-4.286-06	5 01e 07
I CD incineration	Bromomethane	secondary	-4.480-00 -4.59e-06	-5.13e-07
LCD incineration	Nickal	secondary		7.04e.07
LCD incineration	Makukdanum	secondary	-0.296-00	-7.046-07
LCD incineration	Chromium (VI)	secondary	-0.436-00	-7.216-07
Eval Oil #4 Brad	Ammonia	secondary	-0.056-00	-7.446-07
Fuel Oil #4 Prod	Allinoma Uvdragen gulfida	secondary	-7.200-00	-0.120-07
Fuel Oil #4 Prod.	Hydrogen sullae	secondary	-8.41e-00	-9.41e-07
Natural Gas Prod.	Formaldenyde	secondary	-9.1/e-06	-1.03e-06
		secondary	-1.40e-05	-1.566-06
	Zinc (elemental)	secondary	-1.65e-05	-1.84e-06
Fuel Oil #4 Prod.	Fluorides (F-)	secondary	-1./Se-05	-1.95e-06
Fuel Oil #4 Prod.	Selenium	secondary	-1.//e-05	-1.98e-06
Natural Gas Prod.	Hydrochloric acid	secondary	-2.11e-05	-2.37e-06
Natural Gas Prod.	Vanadium	secondary	-2.30e-05	-2.58e-06
Fuel Oil #4 Prod.	Phosphorus (yellow or white)	secondary	-2.63e-05	-2.94e-06
LCD incineration	Ammonia	secondary	-3.04e-05	-3.40e-06
LCD incineration	Phosphorus (yellow or white)	secondary	-3.56e-05	-3.98e-06
Fuel Oil #4 Prod.	Nitrous oxide	secondary	-4.08e-05	-4.56e-06
Fuel Oil #4 Prod.	Hydrochloric acid	secondary	-8.74e-05	-9.77e-06
Natural Gas Prod.	Ammonia	secondary	-8.90e-05	-9.96e-06
Natural Gas Prod.	Sulfur oxides	secondary	-9.13e-05	-1.02e-05
Natural Gas Prod.	Arsenic	secondary	-1.04e-04	-1.17e-05
Natural Gas Prod.	PM	secondary	-1.37e-04	-1.53e-05
LCD incineration	Hydrofluoric acid	secondary	-1.45e-04	-1.62e-05

Process Group	Material	LCI Data Type	Terrestrial Toxicity (tox-kg)	% of Total
Fuel Oil #4 Prod.	PM	secondary	-2.87e-04	-3.21e-05
LCD incineration	Formaldehyde	secondary	-2.89e-04	-3.24e-05
Fuel Oil #4 Prod.	Formaldehyde	secondary	-3.65e-04	-4.08e-05
Fuel Oil #4 Prod.	Arsenic	secondary	-5.00e-04	-5.59e-05
LCD incineration	Nitrous oxide	secondary	-6.75e-04	-7.55e-05
LCD incineration	Vanadium	secondary	-7.59e-04	-8.49e-05
LCD incineration	Selenium	secondary	-9.94e-04	-1.11e-04
LCD incineration	Fluorides (F-)	secondary	-1.01e-03	-1.13e-04
Fuel Oil #4 Prod.	Nitrogen oxides	secondary	-1.27e-03	-1.43e-04
Fuel Oil #4 Prod.	Benzene	secondary	-1.71e-03	-1.91e-04
Fuel Oil #4 Prod.	Sulfur oxides	secondary	-1.74e-03	-1.94e-04
Fuel Oil #4 Prod.	Methane	secondary	-1.85e-03	-2.07e-04
LCD incineration	Benzene	secondary	-2.08e-03	-2.32e-04
Fuel Oil #4 Prod.	Vanadium	secondary	-2.86e-03	-3.20e-04
LCD incineration	Hydrochloric acid	secondary	-3.08e-03	-3.44e-04
Natural Gas Prod.	Nitrogen oxides	secondary	-3.09e-03	-3.45e-04
LCD incineration	Nitrogen oxides	secondary	-9.31e-03	-1.04e-03
LCD incineration	Methane	secondary	-9.85e-03	-1.10e-03
Fuel Oil #4 Prod.	Carbon monoxide	secondary	-9.96e-03	-1.11e-03
LCD incineration	PM	secondary	-1.20e-02	-1.34e-03
LCD incineration	Carbon monoxide	secondary	-1.44e-02	-1.61e-03
Natural Gas Prod.	Methane	secondary	-1.66e-02	-1.86e-03
LCD incineration	Sulfur oxides	secondary	-1.75e-02	-1.96e-03
Natural Gas Prod.	Carbon monoxide	secondary	-2.11e-02	-2.36e-03
LCD incineration	Arsenic	secondary	-2.50e-02	-2.80e-03
Natural Gas Prod.	Benzene	secondary	-2.50e-02	-2.80e-03
Total End-of-life			1.46e-02	1.63e-03
Total All Life-cycle Stages			8.94e+02	1.00e+02

 Table M-40. LCD LCIA Results for the Terrestrial Ecotoxicity Impact Category

this page intentionally left blank

APPENDIX N

CHAPTER 4 SUPPORTING TABLES
this page intentionally left blank

	Table N-1. CKI	Lead Inputs by Life-cycle Stag	ē		
Life-cycle Stage	Process	Input	Quantity	Units	Type
Materials Processing	Aluminum Prod. (all virgin; EB)	Lead (Pb, ore)	6.30e-05	kg	Ancillary material
Materials Processing	Ferrite mfg. (EB)	Lead (Pb, ore)	5.70e-08	kg	Ancillary material
Materials Processing	Invar (DEAM mix)	Lead (Pb, ore)	5.84e-08	kg	Ancillary material
Materials Processing	Lead (EB)	Lead (Pb, ore)	4.96e-01	kg	Primary material
Materials Processing	Polycarbonate Production (PC; DEAM)	Lead (Pb, ore)	1.85e-06	kg	Ancillary material
Materials Processing	Steel Prod., cold-rolled, semi-finished (EB)	Lead (Pb, ore)	5.79e-08	kg	Ancillary material
Manufacturing	CRT tube mfg. (CDP)	Frit	6.67e-02	kg	Primary material
Manufacturing	CRT glass mfg. (CDP)	Lead	4.47e-01	kg	Primary material
Manufacturing	Frit manufacturing (CDP)	Lead	4.67e-02	kg	Primary material
Manufacturing	CRT monitor assembly (CDP)	Printed wiring board (PWB)	8.47e-01	kg	Direct to assembly
Manufacturing	PWB Mfg.	Solder (63% tin; 37% lead)	5.08e-02	kg	Primary material
Manufacturing	CRT monitor assembly (CDP)	Solder, unspecified	2.67e-02	kg	Direct to assembly
End-of-life	CRT Incineration (DEAM mix)	EOL CRT Monitor, incinerated	2.20e+01	kg	Primary material
End-of-life	CRT landfilling (CDP)	EOL CRT Monitor, landfilled	1.56e+01	kg	Primary material
End-of-life	CRT Recycling (CDP)	EOL CRT Monitor, recycled	2.42e+00	kg	Primary material

U. ŕ T :: f. ľ Lac Table N 1

Table N-2. LC	D Lead Inputs by Life-cycle Stag	G			
Life-cycle Stage	Process	Input	Quantity	Units	Type
Materials Processing	Aluminum Prod. (all virgin; EB)	Lead (Pb, ore)	2.35e-05	kg	Ancillary material
Materials Processing	PET Resin Production (DEAM)	Lead (Pb, ore)	1.82e-07	kg	Ancillary material
Materials Processing	Polycarbonate Production (PC; DEAM)	Lead (Pb, ore)	1.03e-06	kg	Ancillary material
Materials Processing	Steel Prod., cold-rolled, semi-finished (EB)	Lead (Pb, ore)	2.84e-08	kg	Ancillary material
Manufacturing	LCD module mfg. (CDP)	Printed wiring board (PWB)	2.27e-02	kg	Direct to assembly
Manufacturing	LCD monitor assembly (CDP)	Printed wiring board (PWB)	3.51e-01	kg	Direct to assembly
Manufacturing	LCD monitor assembly (CDP)	Solder (60% tin, 40% lead)	3.81e-02	kg	Primary material
Manufacturing	PWB Mfg.	Solder (63% tin; 37% lead)	2.24e-02	kg	Primary material
Manufacturing	LCD module mfg. (CDP)	Solder, unspecified	7.35e-05	kg	Ancillary material
End-of-life	LCD incineration (DEAM mix)	EOL LCD Monitor, incinerated	6.50e+00	kg	Primary material
End-of-life	LCD landfilling (CDP)	EOL LCD Monitor, landfilled	3.57e+00	kg	Primary material
End-of-life	LCD recycling (CDP)	EOL LCD Monitor, recycled	9.75e-01	kg	Primary material

APPENDIX N

Life-Cycle Stage	Process	Output	Quantity	Units	Type	Disposition
Materials Processing	ABS Production (DEAM)	Lead	2.12e-07	kg	Airborne	air
Materials Processing	Aluminum Prod. (all virgin; EB)	Lead	1.20e-06	kg	Airborne	air
Materials Processing	Aluminum Prod. (all virgin; EB)	Lead	1.45e-08	kg	Solid waste	landfill
Materials Processing	Aluminum Prod. (all virgin; EB)	Lead cmpds	8.72e-06	kg	Waterborne	surface water
Materials Processing	Ferrite mfg. (EB)	Lead	3.46e-05	kg	Airborne	air
Materials Processing	Ferrite mfg. (EB)	Lead	1.70e-10	kg	Solid waste	landfill
Materials Processing	Ferrite mfg. (EB)	Lead cmpds	5.19e-07	kg	Waterborne	surface water
Materials Processing	Ferrite mfg. (EB)	Lead-210 (isotope)	1.82e-01	Bq	Radioactivity	air
Materials Processing	Invar (DEAM mix)	Lead	3.62e-05	kg	Airborne	air
Materials Processing	Invar (DEAM mix)	Lead	5.49e-09	kg	Solid waste	landfill
Materials Processing	Invar (DEAM mix)	Lead cmpds	2.98e-06	kg	Waterborne	surface water
Materials Processing	Invar (DEAM mix)	Lead-210 (isotope)	1.87e-01	Bq	Radioactivity	air
Materials Processing	Lead (EB)	Lead	1.58e-03	kg	Airborne	air
Materials Processing	Lead (EB)	Lead	2.72e-09	kg	Solid waste	landfill
Materials Processing	Lead (EB)	Lead cmpds	2.75e-06	kg	Waterborne	surface water
Materials Processing	Polycarbonate Production (PC; DEAM)	Lead	4.62e-07	kg	Airborne	air
Materials Processing	Steel Prod., cold-rolled, semi-finished (EB)	Lead	1.10e-06	kg	Airborne	air
Materials Processing	Steel Prod., cold-rolled, semi-finished (EB)	Lead	1.02e-11	kg	Solid waste	landfill
Materials Processing	Steel Prod., cold-rolled, semi-finished (EB)	Lead cmpds	8.83e-07	kg	Waterborne	surface water
Materials Processing	Steel Prod., cold-rolled, semi-finished (EB)	Lead-210 (isotope)	6.54e-01	Bq	Radioactivity	air
Materials Processing	Styrene-butadiene Copolymer Prod. (DEAM mix)	Lead	2.07e-07	kg	Airborne	air
Manufacturing	CRT glass mfg. (CDP)	Broken CRT glass	1.88e-03	kg	Hazardous waste	landfill
Manufacturing	CRT glass mfg. (CDP)	cinders from CRT glass mfg (70% PbO)	8.26e-03	kg	Hazardous waste	landfill
Manufacturing	CRT glass mfg. (CDP)	CRT glass faceplate EP dust (Pb) (D008 waste)	1.03e-03	kg	Hazardous waste	landfill
Manufacturing	CRT glass mfg. (CDP)	CRT glass funnel EP dust (Pb) (D008 waste)	5.01e-03	kg	Hazardous waste	recycling/reuse

Table N-3. CRT Lead Outputs by Life-cycle Stage

	T and T - J.	UNI Frau Outputs by FIIC-CVUC	DLAZU			
Life-Cycle Stage	Process	Output	Quantity	Units	Type	Disposition
Manufacturing	CRT glass mfg. (CDP)	Hazardous sludge (Pb) (D008)	1.52e-03	kg	Hazardous waste	landfill
Manufacturing	CRT glass mfg. (CDP)	Lead	3.22e-07	kg	Airborne	air
Manufacturing	CRT glass mfg. (CDP)	Lead	4.34e-05	kg	Waterborne	surface water
Manufacturing	CRT glass mfg. (CDP)	Lead contaminated grit (D008 waste)	3.46e-05	kg	Hazardous waste	landfill
Manufacturing	CRT glass mfg. (CDP)	Lead debris (D008 waste)	2.14e-04	kg	Hazardous waste	landfill
Manufacturing	CRT glass mfg. (CDP)	sludge from CRT glass mfg (1% PbO)	8.78e-04	kg	Hazardous waste	landfill
Manufacturing	CRT glass mfg. (CDP)	Waste Batch (Ba, Pb) (D008 waste)	1.41e-03	kg	Hazardous waste	landfill
Manufacturing	CRT glass mfg. (CDP)	Waste finishing sludge (Pb) (D008 waste)	2.56e-04	kg	Hazardous waste	landfill
Manufacturing	CRT monitor assembly (CDP)	Broken CRT glass	3.82e-01	kg	Solid waste	recycling/reuse
Manufacturing	CRT monitor assembly (CDP)	Printed wiring board (PWB)	3.70e-02	kg	Solid waste	recycling/reuse
Manufacturing	CRT tube mfg. (CDP)	Broken CRT glass	6.94e-01	kg	Solid waste	recycling/reuse
Manufacturing	CRT tube mfg. (CDP)	Frit	2.99e-03	kg	Hazardous waste	landfill
Manufacturing	CRT tube mfg. (CDP)	Lead	3.01e-06	kg	Waterborne	surface water
Manufacturing	CRT tube mfg. (CDP)	Lead	1.03e-06	kg	Waterborne	treatment
Manufacturing	CRT tube mfg. (CDP)	Lead sulfate cake	2.67e-05	kg	Hazardous waste	landfill
Manufacturing	Frit manufacturing (CDP)	Lead	3.20e-10	kg	Airborne	air
Manufacturing	Fuel Oil #2 Prod. (DEAM)	Lead	3.88e-08	kg	Airborne	air
Manufacturing	Fuel Oil #2 Prod. (DEAM)	Lead cmpds	3.28e-12	kg	Waterborne	surface water
Manufacturing	Fuel Oil #4 Prod. (DEAM mix)	Lead	4.14e-09	kg	Airborne	air
Manufacturing	Fuel Oil #4 Prod. (DEAM mix)	Lead cmpds	2.66e-13	kg	Waterborne	surface water
Manufacturing	Fuel Oil #6 Prod. (DEAM)	Lead	9.99e-08	kg	Airborne	air
Manufacturing	Fuel Oil #6 Prod. (DEAM)	Lead cmpds	3.95e-12	kg	Waterborne	surface water
Manufacturing	Japanese Electric Grid	Lead (Pb, ore)	4.41e-07	kg	Airborne	air
Manufacturing	LPG Production (DEAM)	Lead	1.24e-05	kg	Airborne	air
Manufacturing	LPG Production (DEAM)	Lead cmpds	1.17e-09	kg	Waterborne	surface water

Table N-3. CRT Lead Outputs by Life-cycle Stage

Life-Cycle Stage	Process	Output	Quantity	Units	Type	Disposition
Manufacturing	Natural Gas Prod. (DEAM)	Lead	2.17e-08	kg	Airborne	air
Manufacturing	Natural Gas Prod. (DEAM)	Lead cmpds	2.36e-14	kg	Waterborne	surface water
Manufacturing	PWB Mfg.	Lead cmpds	1.62e-05	kg	Waterborne	treatment
Manufacturing	PWB Mfg.	PWB-Solder dross	6.70e-02	kg	Hazardous waste	recycling/reuse
Manufacturing	US electric grid	Lead	2.04e-07	kg	Airborne	air
Use	CRT monitor use (CDP)	EOL CRT Monitor, landfilled	1.01e+01	kg	Hazardous waste	treatment
Use	CRT monitor use (CDP)	EOL CRT Monitor, landfilled	2.12e+01	kg	Solid waste	treatment
Use	CRT monitor use (CDP)	EOL CRT Monitor, recycled	2.42e+00	kg	Solid waste	treatment
Use	CRT monitor use (CDP)	EOL CRT Monitor, remanufactured	6.60e-01	kg	Solid waste	recycling/reuse
Use	US electric grid	Lead	1.28e-05	kg	Airborne	air
End-of-life	CRT Incineration (DEAM mix)	Lead	1.43e-05	kg	Airborne	air
End-of-life	CRT Incineration (DEAM mix)	Lead cmpds	8.11e-10	kg	Waterborne	surface water
End-of-life	CRT landfilling (CDP)	EOL CRT Monitor, landfilled	3.91e+00	kg	Solid waste	landfill
End-of-life	CRT landfilling (CDP)	Lead cmpds	7.90e-10	kg	Waterborne	surface water
End-of-life	CRT Recycling (CDP)	Printed wiring board (PWB)	1.46e-01	kg	Hazardous waste	recycling/reuse
End-of-life	Fuel Oil #4 Prod. (DEAM mix)	Lead	-4.45e-08	kg	Airborne	air
End-of-life	Fuel Oil #4 Prod. (DEAM mix)	Lead cmpds	-2.86e-12	kg	Waterborne	surface water
End-of-life	LPG Production (DEAM)	Lead	1.07e-10	kg	Airborne	air
End-of-life	LPG Production (DEAM)	Lead cmpds	1.01e-14	kg	Waterborne	surface water
End-of-life	Natural Gas Prod. (DEAM)	Lead	-1.09e-08	kg	Airborne	air
End-of-life	Natural Gas Prod. (DEAM)	Lead cmpds	-1.18e-14	kg	Waterborne	surface water
End-of-life	US electric grid	Lead	1.28e-09	kg	Airborne	air

Table N-3. CRT Lead Outputs by Life-cycle Stage

7
~
-
Ц
Z
Ξ.
A
2
- <⊡
~

	Table N-4. L(CD Lead Outputs by Life-cyc	cle Stage			
Life-cycle Stage	Process	Output	Quantity	Units	Type	Disposition
Materials Processing	Aluminum Prod. (all virgin; EB)	Lead	4.47e-07	kg	Airborne	air
Materials Processing	Aluminum Prod. (all virgin; EB)	Lead	5.41e-09	kg	Solid waste	landfill
Materials Processing	Aluminum Prod. (all virgin; EB)	Lead cmpds	3.25e-06	kg	Waterborne	surface water
Materials Processing	Natural Gas Prod. (DEAM)	Lead	1.51e-06	kg	Airborne	air
Materials Processing	Natural Gas Prod. (DEAM)	Lead cmpds	1.65e-12	kg	Waterborne	surface water
Materials Processing	PET Resin Production (DEAM)	Lead	9.08e-08	kg	Airborne	air
Materials Processing	PMMA Sheet Prod. (DEAM)	Lead	1.92e-07	kg	Airborne	air
Materials Processing	Polycarbonate Production (PC; DEAM)	Lead	2.58e-07	kg	Airborne	air
Materials Processing	Steel Prod., cold-rolled, semi-finished (EB)	Lead	5.40e-07	kg	Airborne	air
Materials Processing	Steel Prod., cold-rolled, semi-finished (EB)	Lead	4.98e-12	kg	Solid waste	landfill
Materials Processing	Steel Prod., cold-rolled, semi-finished (EB)	Lead cmpds	4.33e-07	kg	Waterborne	surface water
Materials Processing	Steel Prod., cold-rolled, semi-finished (EB)	Lead-210 (isotope)	3.21e-01	Bq	Radioactivity	air
Materials Processing	Styrene-butadiene Copolymer Prod. (DEAM mix)	Lead	9.04e-08	kg	Airborne	air
Manufacturing	Fuel Oil #2 Prod. (DEAM)	Lead	1.81e-09	kg	Airborne	air
Manufacturing	Fuel Oil #2 Prod. (DEAM)	Lead cmpds	1.53e-13	kg	Waterborne	surface water
Manufacturing	Fuel Oil #4 Prod. (DEAM mix)	Lead	6.38e-09	kg	Airborne	air
Manufacturing	Fuel Oil #4 Prod. (DEAM mix)	Lead cmpds	4.10e-13	kg	Waterborne	surface water
Manufacturing	Fuel Oil #6 Prod. (DEAM)	Lead	3.40e-09	kg	Airborne	air
Manufacturing	Fuel Oil #6 Prod. (DEAM)	Lead cmpds	1.34e-13	kg	Waterborne	surface water
Manufacturing	Japanese Electric Grid	Lead (Pb, ore)	1.48e-06	kg	Airborne	air
Manufacturing	LCD backlight unit assembly (CDP)	Waste CCFL, with lead	8.17e-08	kg	Hazardous waste	treatment
Manufacturing	LCD CCFL mfg. (CDP)	Lead	8.33e-07	kg	Waterborne	treatment
Manufacturing	LCD glass mfg. (CDP)	Lead	2.01e-06	kg	Waterborne	surface water
Manufacturing	LCD glass mfg. (CDP)	Waste Batch (Ba, Pb) (D008 waste)	6.55e-05	kg	Hazardous waste	landfill

9-V

	Table N-4. L(CD Lead Outputs by Life-cyd	cle Stage			
Life-cycle Stage	Process	Output	Quantity	Units	Type	Disposition
Manufacturing	LCD module mfg. (CDP)	Lead	6.17e-06	kg	Waterborne	surface water
Manufacturing	LCD monitor assembly (CDP)	Printed wiring board (PWB)	7.50e-03	kg	Solid waste	landfill
Manufacturing	LPG Production (DEAM)	Lead	5.93e-07	kg	Airborne	air
Manufacturing	LPG Production (DEAM)	Lead cmpds	5.60e-11	kg	Waterborne	surface water
Manufacturing	Natural Gas Prod. (DEAM)	Lead	3.42e-08	kg	Airborne	air
Manufacturing	Natural Gas Prod. (DEAM)	Lead cmpds	3.72e-14	kg	Waterborne	surface water
Manufacturing	PWB Mfg.	Lead cmpds	7.14e-06	kg	Waterborne	treatment
Manufacturing	PWB Mfg.	PWB-Lead contaminated waste oil	5.14e-03	kg	Hazardous waste	treatment
Manufacturing	PWB Mfg.	PWB-Solder dross	2.96e-02	kg	Hazardous waste	recycling/reuse
Manufacturing	US electric grid	Lead	2.47e-08	kg	Airborne	air
Use	LCD monitor use (CDP)	EOL LCD Monitor, incinerated	9.75e-01	kg	Solid waste	treatment
Use	LCD monitor use (CDP)	EOL LCD Monitor, landfilled	3.25e-01	kg	Hazardous waste	treatment
Use	LCD monitor use (CDP)	EOL LCD Monitor, landfilled	6.82e+00	kg	Solid waste	treatment
Use	LCD monitor use (CDP)	EOL LCD Monitor, recycled	9.75e-01	kg	Solid waste	treatment
Use	LCD monitor use (CDP)	EOL LCD Monitor, remanufactured	9.75e-01	kg	Solid waste	recycling/reuse
Use	US electric grid	Lead	4.76e-06	kg	Airborne	air
End-of-life	Fuel Oil #4 Prod. (DEAM mix)	Lead	-2.90e-08	kg	Airborne	air
End-of-life	Fuel Oil #4 Prod. (DEAM mix)	Lead cmpds	-1.90e-12	kg	Waterborne	surface water
End-of-life	LCD incineration (DEAM mix)	Lead	4.80e-06	kg	Airborne	air
End-of-life	LCD incineration (DEAM mix)	Lead cmpds	2.59e-10	kg	Waterborne	surface water
End-of-life	LCD landfilling (CDP)	EOL LCD Monitor, landfilled	8.94e-01	kg	Solid waste	landfill
End-of-life	LCD landfilling (CDP)	EOL LCD Monitor, landfilled	1.64e+00	kg	Hazardous waste	landfill
End-of-life	LCD landfilling (CDP)	Lead cmpds	2.40e-10	kg	Waterborne	surface water
End-of-life	LPG Production (DEAM)	Lead	4.87e-11	kg	Airborne	air
End-of-life	LPG Production (DEAM)	Lead cmpds	4.60e-15	kg	Waterborne	surface water

É T : L Þ Ę Þ . 7

L-N

by Life-cycle Stage	
Lead Outputs	
Table N-4. LCD	

Life-cycle Stage	Process	Output	Quantity	Units	Type	Disposition
End-of-life	Natural Gas Prod. (DEAM)	Lead	-7.20e-09	kg	Airborne	air
End-of-life	Natural Gas Prod. (DEAM)	Lead cmpds	-7.80e-15	kg	Waterborne	surface water
End-of-life	US electric grid	Lead	9.03e-10	kg	Airborne	air

Table N-5. CRT Life-Cycle Impacts Scores from Lead-based Materials

Life-cycle	Process	Input/Output			Impact	Scores by Cat	egory		
Stage			Nonrenew.	Haz. Waste	Rad	ChrTox- pub	ChrTox- occ	Aq. Tox.	Terr. Tox.
MP	ABS Production (DEAM)	Lead				4.24e-07			2.12e-07
MP	Aluminum Prod. (all virgin; EB)	Lead				2.40e-06			1.20e-06
MP	Ferrite mfg. (EB)	Lead				6.92e-05			3.46e-05
MP	Invar (DEAM mix)	Lead				7.24e-05			3.62e-05
MP	Lead (EB)	Lead				3.16e-03			1.58e-03
MP	Polycarbonate Production (PC; DEAM)	Lead				9.23e-07			4.62e-07
MP	Steel Prod., cold-rolled, semi-finished (EB)	Lead				2.20e-06			1.10e-06
MP	Styrene-butadiene Copolymer Prod. (DEAM	Lead				4.14e-07			2.07e-07
MP	Aluminum Prod. (all virgin; EB)	Lead (Pb, ore)	6.30e-05						
MP	Ferrite mfg. (EB)	Lead (Pb, ore)	5.70e-08						
MP	Invar (DEAM mix)	Lead (Pb, ore)	5.84e-08						
MP	Lead (EB)	Lead (Pb, ore)	4.96e-01						
MP	Polycarbonate Production (PC; DEAM)	Lead (Pb, ore)	1.85e-06						
MP	Steel Prod., cold-rolled, semi-finished (EB)	Lead (Pb, ore)	5.79e-08						
MP	Aluminum Prod. (all virgin; EB)	Lead cmpds				1.74e-05		1.70e-04	8.72e-06
MP	Ferrite mfg. (EB)	Lead cmpds	1.04e-06	1.00e-05	5.19e-07				
MP	Invar (DEAM mix)	Lead cmpds				5.96e-06		5.90e-05	2.98e-06

APPENDIX N

		NT FILC-CYCIC TIIIDAN	CO TOOO CO	II OIII TCAL	I-Nasca I	ATAUAT JAIN			
Life-cycle	Process	Input/Output			Impact	Scores by Cat	tegory		
Stage			Nonrenew.	Haz. Waste	Rad	ChrTox- pub	ChrTox- occ	Aq. Tox.	Terr. Tox.
MP	Lead (EB)	Lead cmpds				5.50e-06		5.40e-05	2.75e-06
MP	Steel Prod., cold-rolled, semi-finished (EB)	Lead cmpds				1.77e-06		1.70e-05	8.83e-07
MP	Ferrite mfg. (EB)	Lead-210 (isotope)			1.82e-01				
MP	Invar (DEAM mix)	Lead-210 (isotope)			1.87e-01				
MP	Steel Prod., cold-rolled, semi-finished (EB)	Lead-210 (isotope)			6.54e-01				
Manufacturing	CRT glass mfg. (CDP)	Broken CRT glass		6.22e-07					
Manufacturing	CRT glass mfg. (CDP)	cinders from CRT glass mfg (70% PbO)		6.88e-06					
Manufacturing	CRT glass mfg. (CDP)	CRT glass faceplate EP dust (Pb) (D008 waste)		2.15e-06					
Manufacturing	CRT tube mfg. (CDP)	Frit		3.04e-06					
Manufacturing	CRT glass mfg. (CDP)	Hazardous sludge (Pb)		1.38e-06					
Manufacturing	CRT glass mfg. (CDP)	Lead	4.47e-01			8.74e-05	8.95e-01	8.70e-05	4.37e-05
Manufacturing	CRT tube mfg. (CDP)	Lead				6.03e-06		6.00e-06	3.01e-06
Manufacturing	Frit manufacturing (CDP)	Lead	4.67e-02			6.40e-10	9.33e-02		3.20e-10
Manufacturing	Fuel Oil #2 Prod. (DEAM)	Lead				7.76e-08			3.88e-08
Manufacturing	Fuel Oil #4 Prod. (DEAM mix)	Lead				8.28e-09			4.14e-09
Manufacturing	Fuel Oil #6 Prod. (DEAM)	Lead				2.00e-07			9.99e-08
Manufacturing	LPG Production (DEAM)	Lead				2.47e-05			1.24e-05
Manufacturing	Natural Gas Prod. (DEAM)	Lead				4.33e-08			2.17e-08
Manufacturing	US electric grid	Lead				4.07e-07			2.04e-07
Manufacturing	Fuel Oil #2 Prod. (DEAM)	Lead cmpds				6.55e-12		6.50e-11	3.27e-12
Manufacturing	Fuel Oil #4 Prod. (DEAM mix)	Lead cmpds		5.32e-13	5.2E-				
Manufacturing	Fuel Oil #6 Prod. (DEAM)	Lead cmpds				7.89e-12		7.80e-11	3.95e-12
Manufacturing	L.P.G. Production (DEAM)	Lead cmnds				2.33e-09		2.30e-08	1.17e-09

Table N-5. CRT Life-Cycle Impacts Scores from Lead-based Materials

Z
ENDIX
APP

Life-cycle	Process	Input/Output			Impact	Scores by Cat	egory		
Stage			Nonrenew.	Haz. Waste	Rad	ChrTox- pub	ChrTox- occ	Aq. Tox.	Terr. Tox.
Manufacturing	Natural Gas Prod. (DEAM)	Lead cmpds				4.72e-14		4.60e-13	2.36e-14
Manufacturing	CRT glass mfg. (CDP)	Lead contaminated grit (D008 waste)		2.99e-09					
Manufacturing	CRT glass mfg. (CDP)	Lead debris (D008 waste)		1.85e-08					
Manufacturing	CRT tube mfg. (CDP)	Lead sulfate cake		3.03e-08					
Manufacturing	CRT glass mfg. (CDP)	sludge from CRT glass mfg (1% PbO)		6.45e-07					
Manufacturing	CRT glass mfg. (CDP)	Waste Batch (Ba, Pb) (D008 waste)		1.22e-07					
Manufacturing	CRT glass mfg. (CDP)	Waste finishing sludge (Pb) (D008 waste)		2.32e-07					
Use	US electric grid	Lead				2.55e-05			1.27e-05
End-of-life	CRT Incineration (DEAM mix)	Lead		2.86e-05	1.43e-05				
End-of-life	Fuel Oil #4 Prod. (DEAM mix)	Lead				-8.90e-08			-4.40e-08
End-of-life	LPG Production (DEAM)	Lead				2.13e-10			1.07e-10
End-of-life	Natural Gas Prod. (DEAM)	Lead				-2.17e-08			-1.10e-08
End-of-life	US electric grid	Lead				2.55e-09			1.28e-09
End-of-life	CRT Incineration (DEAM mix)	Lead empds				1.62e-09		1.60e-08	8.10e-10
End-of-life	CRT landfilling (CDP)	Lead empds				1.58e-09		1.60e-08	7.89e-10
End-of-life	Fuel Oil #4 Prod. (DEAM mix)	Lead cmpds				-5.72e-12		-5.60e-11	-2.90e-12
End-of-life	LPG Production (DEAM)	Lead cmpds				2.01e-14		2.00e-13	1.01e-14
End-of-life	Natural Gas Prod. (DEAM)	Lead cmpds				-2.37e-14		-2.30e-13	-1.20e-14
MP = Materials pr	ocessing.								

Blank = not applicable.

APPENDIX N

Life-cycle Stage	Process	Input/Output			Impact Sco	res by Categ	gory		
			Nonrenew	Haz. Waste	Solid Waste	Rad	ChrTox- pub	Aq. Tox.	Terr. Tox.
MP	Aluminum Prod. (all virgin; EB)	Lead					8.94e-07		4.47e-07
MP	Natural Gas Prod. (DEAM)	Lead					3.03e-06		1.51e-06
MP	PET Resin Production (DEAM)	Lead					1.82e-07		9.08e-08
MP	PMMA Sheet Prod. (DEAM)	Lead					3.83e-07		1.92e-07
MP	Polycarbonate Production (PC; DEAM)	Lead					5.16e-07		2.58e-07
MP	Steel Prod., cold-rolled, semi-finished (EB)	Lead					1.08e-06		5.40e-07
MP	Styrene-butadiene Copolymer Prod. (DEAM mix)	Lead					1.81e-07		9.04e-08
MP	Aluminum Prod. (all virgin; EB)	Lead (Pb, ore)	2.35e-05						
MP	PET Resin Production (DEAM)	Lead (Pb, ore)	1.82e-07						
MP	Polycarbonate Production (PC; DEAM)	Lead (Pb, ore)	1.03e-06						
MP	Steel Prod., cold-rolled, semi-finished (EB)	Lead (Pb, ore)	2.84e-08						
MP	Aluminum Prod. (all virgin; EB)	Lead cmpds					6.49e-06	6.40e-05	3.25e-06
MP	Natural Gas Prod. (DEAM)	Lead cmpds					3.30e-12	3.25e-11	1.65e-12
MP	Steel Prod., cold-rolled, semi-finished (EB)	Lead cmpds					8.65e-07	8.52e-06	4.33e-07
MP	Steel Prod., cold-rolled, semi-finished (EB)	Lead-210 (isotope)				3.21e-01			
Manufacturing	Fuel Oil #2 Prod. (DEAM)	Lead					3.62e-09		1.81e-09
Manufacturing	Fuel Oil #4 Prod. (DEAM mix)	Lead					1.28e-08		6.38e-09
Manufacturing	Fuel Oil #6 Prod. (DEAM)	Lead					6.80e-09		3.40e-09
Manufacturing	LCD glass mfg. (CDP)	Lead					4.02e-06	4.02e-06	2.01e-06
Manufacturing	LCD module mfg. (CDP)	Lead					1.23e-05	1.23e-05	6.17e-06
Manufacturing	LPG Production (DEAM)	Lead					1.19e-06		5.93e-07
Manufacturing	Natural Gas Prod. (DEAM)	Lead					6.83e-08		3.42e-08
Manufacturing	US electric grid	Lead					4.94e-08		2.47e-08
Manufacturing	Fuel Oil #2 Prod. (DEAM)	Lead cmpds					3.06e-13	3.01e-12	1.53e-13

Table N-6. LCD Life-Cycle Impact Scores from Lead-based Materials

Z
X
è
E.
P
₹.

	Table N-6. LCL) Life-Cvcle Impac	ct Scores fi	rom Lead-	-based Mat	terials			
Life-cycle Stage	Process	Input/Output			Impact Sco	res by Cate	gory		
			Nonrenew	Haz. Waste	Solid Waste	Rad	ChrTox- pub	Aq. Tox.	Terr. Tox.
Manufacturing	Fuel Oil #4 Prod. (DEAM mix)	Lead cmpds					8.20e-13	8.08e-12	4.10e-13
Manufacturing	Fuel Oil #6 Prod. (DEAM)	Lead cmpds					2.69e-13	2.65e-12	1.34e-13
Manufacturing	LPG Production (DEAM)	Lead cmpds					1.12e-10	1.10e-09	5.60e-11
Manufacturing	Natural Gas Prod. (DEAM)	Lead cmpds					7.44e-14	7.33e-13	3.72e-14
Manufacturing	LCD monitor assembly (CDP)	Printed wiring board			9.38e-06				
Manufacturing	LCD glass mfg. (CDP)	Waste batch (Ba, Pb) (D008 waste)		5.67e-09					
Use	US electric grid	Lead					9.52e-06		4.76e-06
End-of Life	Fuel Oil #4 Prod. (DEAM mix)	Lead					-5.88e-08		-2.94e-08
End-of Life	LCD incineration (DEAM mix)	Lead					9.60e-06		4.80e-06
End-of Life	LPG Production (DEAM)	Lead					9.74e-11		4.87e-11
End-of Life	Natural Gas Prod. (DEAM)	Lead					-1.43e-08		-7.17e-09
End-of Life	US electric grid	Lead					1.81e-09		9.03e-10
End-of Life	Fuel Oil #4 Prod. (DEAM mix)	Lead cmpds					-3.78e-12	-3.72e-11	-1.89e-12
End-of Life	LCD incineration (DEAM mix)	Lead cmpds					5.19e-10	5.11e-09	2.59e-10
End-of Life	LCD landfilling (CDP)	Lead cmpds					4.80e-10	4.73e-09	2.40e-10
End-of Life	LPG Production (DEAM)	Lead empds					9.19e-15	9.06e-14	4.60e-15
End-of Life	Natural Gas Prod. (DEAM)	Lead empds					-1.56e-14	-1.54e-13	-7.81e-15
$M = M_{abarballa}$									

¢ **U**+ Ć 4 4 ζ 2 Ę Ē

MP = Materials processing. Blank = not applicable.

Table N-7. LCD Mercury Inputs by Life-cycle Stage

Life-cycle Stage	Process	Input	Quantity	Units	Type
Manufacturing	LCD CCFL mfg. (CDP)	Mercury	3.99e-06	kg	Primary material
Manufacturing	LCD backlight unit assembly (CDP)	Backlight lamp (CCFL)	1.94e-03	kg	Direct to assembly

Z
X
Ē
E
AP

	Table N-8. CRT Mercu	irv Outputs by Life-cyc	le Stage			
Life-cycle Stage	Process	Output	Quantity	Units	Type	Disposition
Materials Processing	ABS Production (DEAM)	Mercury	2.12e-07	kg	Airborne	air
Materials Processing	ABS Production (DEAM)	Mercury compounds	2.12e-07	kg	Waterborne	surface water
Materials Processing	Aluminum Prod. (all virgin; EB)	Mercury	2.02e-07	kg	Airborne	air
Materials Processing	Aluminum Prod. (all virgin; EB)	Mercury	8.93e-11	kg	Solid waste	landfill
Materials Processing	Ferrite mfg. (EB)	Mercury	1.62e-07	kg	Airborne	air
Materials Processing	Ferrite mfg. (EB)	Mercury	1.11e-12	kg	Solid waste	landfill
Materials Processing	Ferrite mfg. (EB)	Mercury compounds	1.03e-10	kg	Waterborne	surface water
Materials Processing	Invar (DEAM mix)	Mercury	2.12e-07	kg	Airborne	air
Materials Processing	Invar (DEAM mix)	Mercury	3.39e-11	kg	Solid waste	landfill
Materials Processing	Invar (DEAM mix)	Mercury compounds	1.06e-10	kg	Waterborne	surface water
Materials Processing	Lead (EB)	Mercury	1.51e-06	kg	Airborne	air
Materials Processing	Lead (EB)	Mercury	1.72e-11	kg	Solid waste	landfill
Materials Processing	Lead (EB)	Mercury compounds	1.50e-09	kg	Waterborne	surface water
Materials Processing	Polycarbonate Production (PC; DEAM)	Mercury	4.62e-07	kg	Airborne	air
Materials Processing	Polycarbonate Production (PC; DEAM)	Mercury compounds	4.62e-07	kg	Waterborne	surface water
Materials Processing	Steel Prod., cold-rolled, semi-finished (EB)	Mercury	3.26e-08	kg	Airborne	air
Materials Processing	Steel Prod., cold-rolled, semi-finished (EB)	Mercury	8.26e-14	kg	Solid waste	landfill
Materials Processing	Steel Prod., cold-rolled, semi-finished (EB)	Mercury compounds	8.58e-08	kg	Waterborne	surface water
Materials Processing	Styrene-butadiene Copolymer Prod. (DEAM mix)	Mercury	2.07e-07	kg	Airborne	air
Materials Processing	Styrene-butadiene Copolymer Prod. (DEAM mix)	Mercury compounds	2.07e-07	kg	Waterborne	surface water
Manufacturing	Fuel Oil #2 Prod. (DEAM)	Mercury	2.72e-09	kg	Airborne	air
Manufacturing	Fuel Oil #2 Prod. (DEAM)	Mercury compounds	3.77e-15	kg	Waterborne	surface water
Manufacturing	Fuel Oil #4 Prod. (DEAM mix)	Mercury	2.89e-10	kg	Airborne	air
Manufacturing	Fuel Oil #4 Prod. (DEAM mix)	Mercury compounds	3.06e-16	kg	Waterborne	surface water
Manufacturing	Fuel Oil #6 Prod. (DEAM)	Mercury	6.94e-09	kg	Airborne	air

APPENDIX N

	I able N-8. CKI Mercu	iry Outputs by Lite-cyci	le Stage			
Life-cycle Stage	Process	Output	Quantity	Units	Type	Disposition
Manufacturing	Fuel Oil #6 Prod. (DEAM)	Mercury compounds	4.54e-15	kg	Waterborne	surface water
Manufacturing	Japanese Electric Grid	Mercury	1.18e-07	kg	Airborne	air
Manufacturing	LPG Production (DEAM)	Mercury	8.69e-07	kg	Airborne	air
Manufacturing	LPG Production (DEAM)	Mercury compounds	1.34e-12	kg	Waterborne	surface water
Manufacturing	Natural Gas Prod. (DEAM)	Mercury	1.88e-09	kg	Airborne	air
Manufacturing	Natural Gas Prod. (DEAM)	Mercury compounds	2.71e-17	kg	Waterborne	surface water
Manufacturing	US electric grid	Mercury	1.20e-07	kg	Airborne	air
Use	US electric grid	Mercury	7.51e-06	kg	Airborne	air
End-of-life	CRT Incineration (DEAM mix)	Mercury	-1.12e-07	kg	Airborne	air
End-of-life	CRT Incineration (DEAM mix)	Mercury compounds	2.15e-11	kg	Waterborne	surface water
End-of-life	CRT landfilling (CDP)	Mercury compounds	2.19e-11	kg	Waterborne	surface water
End-of-life	Fuel Oil #4 Prod. (DEAM mix)	Mercury	-3.11e-09	kg	Airborne	air
End-of-life	Fuel Oil #4 Prod. (DEAM mix)	Mercury compounds	-3.29e-15	kg	Waterborne	surface water
End-of-life	LPG Production (DEAM)	Mercury	7.50e-12	kg	Airborne	air
End-of-life	LPG Production (DEAM)	Mercury compounds	1.16e-17	kg	Waterborne	surface water
End-of-life	Natural Gas Prod. (DEAM)	Mercury	-9.44e-10	kg	Airborne	air
End-of-life	Natural Gas Prod. (DEAM)	Mercury compounds	-1.36e-17	kg	Waterborne	surface water
End-of-life	US electric grid	Mercury	7.52e-10	kg	Airborne	air

Jo Ste I if. Ē --Ċ Tabla N-8 CDT Ma

ł	Z
Ì	X
	2
į	PE)
į	AP
	7

ge	
le Stag	ĺ
CVC	ŀ
Life-	
þ	ŀ
puts	
/ Out	
Mercury	
B	
Ľ	
N-9.	
Table N	
	I

Life-Cycle Stage	Process	Output	Quantity	Units	Type	Disposition
MP	Aluminum Prod. (all virgin; EB)	Mercury	7.51e-08	kg	Airborne	air
MP	Natural Gas Prod. (DEAM)	Mercury	1.32e-07	kg	Airborne	air
MP	PET Resin Production (DEAM)	Mercury	9.08e-08	kg	Airborne	air
MP	PMMA Sheet Prod. (DEAM)	Mercury	1.92e-07	kg	Airborne	air
MP	Polycarbonate Production (PC; DEAM)	Mercury	2.58e-07	kg	Airborne	air
MP	Steel Prod., cold-rolled, semi-finished (EB)	Mercury	1.60e-08	kg	Airborne	air
MP	Styrene-butadiene Copolymer Prod. (DEAM mix)	Mercury	9.04e-08	kg	Airborne	air
MP	Aluminum Prod. (all virgin; EB)	Mercury	3.32e-11	kg	Solid waste	landfill
MP	Steel Prod., cold-rolled, semi-finished (EB)	Mercury	4.05e-14	kg	Solid waste	landfill
MP	PET Resin Production (DEAM)	Mercury	9.08e-08	kg	Waterborne	surface water
MP	Natural Gas Prod. (DEAM)	Mercury compounds	1.89e-15	kg	Waterborne	surface water
MP	PMMA Sheet Prod. (DEAM)	Mercury compounds	1.92e-07	kg	Waterborne	surface water
MP	Polycarbonate Production (PC; DEAM)	Mercury compounds	2.58e-07	kg	Waterborne	surface water
MP	Steel Prod., cold-rolled, semi-finished (EB)	Mercury compounds	4.21e-08	kg	Waterborne	surface water
MP	Styrene-butadiene Copolymer Prod. (DEAM mix)	Mercury compounds	9.04e-08	kg	Waterborne	surface water
Manufacturing	LCD CCFL mfg. (CDP)	Backlight lamp (CCFL)	1.94e-03	kg	Product	
Manufacturing	LCD backlight unit assembly (CDP)	Broken CCFL	2.69e-07	kg	Solid waste	landfill
Manufacturing	Fuel Oil #2 Prod. (DEAM)	Mercury	1.27e-10	kg	Airborne	air
Manufacturing	Fuel Oil #4 Prod. (DEAM mix)	Mercury	4.46e-10	kg	Airborne	air
Manufacturing	Fuel Oil #6 Prod. (DEAM)	Mercury	2.36e-10	kg	Airborne	air
Manufacturing	Japanese Electric Grid	Mercury	3.97e-07	kg	Airborne	air
Manufacturing	LPG Production (DEAM)	Mercury	4.17e-08	kg	Airborne	air
Manufacturing	Natural Gas Prod. (DEAM)	Mercury	2.97e-09	kg	Airborne	air
Manufacturing	US electric grid	Mercury	1.45e-08	kg	Airborne	air
Manufacturing	LCD monitor assembly (CDP)	Mercury	2.00e-06	kg	Hazardous waste	recycling/reuse

		D Mercury Outputs by Lite-cycl	le olage			
Life-Cycle Stage	Process	Output	Quantity	Units	Type	Disposition
Manufacturing	LCD module mfg. (CDP)	Mercury	9.69e-08	kg	Waterborne	surface water
Manufacturing	LCD CCFL mfg. (CDP)	Mercury	8.33e-08	kg	Waterborne	treatment
Manufacturing	Fuel Oil #2 Prod. (DEAM)	Mercury compounds	1.76e-16	kg	Waterborne	surface water
Manufacturing	Fuel Oil #4 Prod. (DEAM mix)	Mercury compounds	4.71e-16	kg	Waterborne	surface water
Manufacturing	Fuel Oil #6 Prod. (DEAM)	Mercury compounds	1.55e-16	kg	Waterborne	surface water
Manufacturing	LPG Production (DEAM)	Mercury compounds	6.44e-14	kg	Waterborne	surface water
Manufacturing	Natural Gas Prod. (DEAM)	Mercury compounds	4.28e-17	kg	Waterborne	surface water
Manufacturing	LCD backlight unit assembly (CDP)	Waste CCFL, with mercury	8.17e-10	kg	Hazardous waste	treatment
Manufacturing	LCD backlight unit assembly (CDP)	Waste glass, with mercury	1.05e-10	kg	Hazardous waste	landfill
Manufacturing	LCD CCFL mfg. (CDP)	Wastewater stream, from CCFL mfg.	1.67e+02	kg	Waterborne	treatment
Use	US electric grid	Mercury	2.80e-06	kg	Airborne	air
End-of-life	Fuel Oil #4 Prod. (DEAM mix)	Mercury	-2.05e-09	kg	Airborne	air
End-of-life	LCD incineration (DEAM mix)	Mercury	-8.42e-08	kg	Airborne	air
End-of-life	LPG Production (DEAM)	Mercury	3.42e-12	kg	Airborne	air
End-of-life	Natural Gas Prod. (DEAM)	Mercury	-6.23e-10	kg	Airborne	air
End-of-life	US electric grid	Mercury	5.32e-10	kg	Airborne	air
End-of-life	Fuel Oil #4 Prod. (DEAM mix)	Mercury compounds	-2.17e-15	kg	Waterborne	surface water
End-of-life	LCD incineration (DEAM mix)	Mercury compounds	8.67e-12	kg	Waterborne	surface water
End-of-life	LCD landfilling (CDP)	Mercury compounds	7.50e-12	kg	Waterborne	surface water
End-of-life	LPG Production (DEAM)	Mercury compounds	5.29e-18	kg	Waterborne	surface water
End-of-life	Natural Gas Prod. (DEAM)	Mercury compounds	-8.98e-18	kg	Waterborne	surface water

Table N-9. LCD Mercury Outputs by Life-cycle Stage

MP = Materials processing

Z	
X	
Ē	
E	i
PP	(
	l

	Table N-10. CRT Life-C	vcle Impact Scores	s trom Me	ercurv-bas	sed Mater	ials		
Life-cycle Stage	Process	Input/Output		a	Impact Scores	by Category		
			Haz. Waste	Solid Waste	ChrTox-pub	ChrTox-occ	Aq. Tox.	Terr. Tox.
MP	ABS Production (DEAM)	Mercury			2.12e-07			2.12e-07
MP	Aluminum Prod. (all virgin; EB)	Mercury			2.02e-07			2.02e-07
MP	Ferrite mfg. (EB)	Mercury			1.62e-07			1.62e-07
MP	Invar (DEAM mix)	Mercury			2.12e-07			2.12e-07
MP	Lead (EB)	Mercury			1.51e-06			1.51e-06
MP	Polycarbonate Production (PC; DEAM)	Mercury			4.62e-07			4.62e-07
MP	Steel Prod., cold-rolled, semi-finished (EB)	Mercury			3.26e-08			3.26e-08
MP	Styrene-butadiene Copolymer Prod. (DEAM mix)	Mercury			2.07e-07			2.07e-07
MP	ABS Production (DEAM)	Mercury compounds			1.12e-04		1.97e-04	1.11e-04
MP	Ferrite mfg. (EB)	Mercury compounds			5.45e-08		9.63e-08	5.44e-08
MP	Invar (DEAM mix)	Mercury compounds			5.58e-08		9.86e-08	5.57e-08
MP	Lead (EB)	Mercury compounds			7.89e-07		1.39e-06	7.88e-07
MP	Polycarbonate Production (PC; DEAM)	Mercury compounds			2.44e-04		4.30e-04	2.43e-04
MP	Steel Prod., cold-rolled, semi-finished (EB)	Mercury compounds			4.53e-05		8.00e-05	4.52e-05
MP	Styrene-butadiene Copolymer Prod. (DEAM mix)	Mercury compounds			1.09e-04		1.93e-04	1.09e-04
Manufacturing	Fuel Oil #2 Prod. (DEAM)	Mercury			2.72e-09			2.72e-09
Manufacturing	Fuel Oil #4 Prod. (DEAM mix)	Mercury			2.89e-10			2.89e-10
Manufacturing	Fuel Oil #6 Prod. (DEAM)	Mercury			6.94e-09			6.94e-09
Manufacturing	Japanese Electric Grid	Mercury			1.18e-07			1.18e-07
Manufacturing	LPG Production (DEAM)	Mercury			8.69e-07			8.69e-07
Manufacturing	Natural Gas Prod. (DEAM)	Mercury			1.88e-09			1.88e-09
Manufacturing	US electric grid	Mercury			1.20e-07			1.20e-07
Manufacturing	Fuel Oil #2 Prod. (DEAM)	Mercury compounds			1.99e-12		3.51e-12	1.98e-12
Manufacturing	Fuel Oil #4 Prod. (DEAM mix)	Mercury compounds			1.61e-13		2.85e-13	1.61e-13

¢ ŭ E

APPENDIX N

	I able N-10. CK1 Lile-C	vole impact scores	S IFOIII ME	rcurv-Da	seu Maler	SIRI		
Life-cycle Stage	Process	Input/Output		•	Impact Score	s by Category		
			Haz. Waste	Solid Waste	ChrTox-pub	ChrTox-occ	Aq. Tox.	Terr. Tox.
Manufacturing	Fuel Oil #6 Prod. (DEAM)	Mercury compounds			2.39e-12		4.23e-12	2.39e-12
Manufacturing	LPG Production (DEAM)	Mercury compounds			7.08e-10		1.25e-09	7.07e-10
Manufacturing	Natural Gas Prod. (DEAM)	Mercury compounds			1.43e-14		2.53e-14	1.43e-14
Use	US electric grid	Mercury			7.51e-06			7.51e-06
End-of-life	CRT Incineration (DEAM mix)	Mercury			-1.12e-07			-1.12e-07
End-of-life	Fuel Oil #4 Prod. (DEAM mix)	Mercury			-3.11e-09			-3.11e-09
End-of-life	LPG Production (DEAM)	Mercury			7.50e-12			7.50e-12
End-of-life	Natural Gas Prod. (DEAM)	Mercury			-9.44e-10			-9.44e-10
End-of-life	US electric grid	Mercury			7.52e-10			7.52e-10
End-of-life	CRT Incineration (DEAM mix)	Mercury compounds			1.13e-08		2.00e-08	1.13e-08
End-of-life	CRT landfilling (CDP)	Mercury compounds			1.15e-08		2.04e-08	1.15e-08
End-of-life	Fuel Oil #4 Prod. (DEAM mix)	Mercury compounds			-1.74e-12		-3.06e-12	-1.73e-12
End-of-life	LPG Production (DEAM)	Mercury compounds			6.11e-15		1.08e-14	6.10e-15
End-of-life	Natural Gas Prod. (DEAM)	Mercury compounds			-7.18e-15		-1.27e-14	-7.16e-15

sed Materials 4 m Ma oc fro ant Con Tahla N-10 CRT Life-Cvela Im

MP = Materials processing. Blank = not applicable.

v-based Materials	
Mercury	
es from	
act Scor	
cle Imp	•
Š	•
Life-	
LCD	
N-11.	
Table N	

Life-cycle Stage	Process	Input/Output			Impact Score	by Category		
			Haz. Waste	Solid Waste	ChrTox- pub	ChrTox- occ	Aq. Tox.	Terr. Tox.
MP	Aluminum Prod. (all virgin; EB)	Mercury			7.51e-08			7.51e-08
MP	Natural Gas Prod. (DEAM)	Mercury			1.32e-07			1.32e-07
MP	PET Resin Production (DEAM)	Mercury			1.82e-07		1.82e-07	1.82e-07
MP	PMMA Sheet Prod. (DEAM)	Mercury			1.92e-07			1.92e-07
MP	Polycarbonate Production (PC; DEAM)	Mercury			2.58e-07			2.58e-07
MP	Steel Prod., cold-rolled, semi-finished (EB)	Mercury			1.60e-08			1.60e-08
MP	Styrene-butadiene Copolymer Prod. (DEAM mix)	Mercury			9.04e-08			9.04e-08
MP	Natural Gas Prod. (DEAM)	Mercury compounds			1.00e-12		1.77e-12	9.98e-13
MP	PMMA Sheet Prod. (DEAM)	Mercury compounds			1.01e-04		1.79e-04	1.01e-04
MP	Polycarbonate Production (PC; DEAM)	Mercury compounds			1.36e-04		2.40e-04	1.36e-04
MP	Steel Prod., cold-rolled, semi-finished (EB)	Mercury compounds			2.22e-05		3.92e-05	2.22e-05
MP	Styrene-butadiene Copolymer Prod. (DEAM mix)	Mercury compounds			4.77e-05		8.42e-05	4.76e-05
Manufacturing	LCD backlight unit assembly (CDP)	Broken CCFL		1.98e-11				
Manufacturing	Fuel Oil #2 Prod. (DEAM)	Mercury			1.27e-10			1.27e-10
Manufacturing	Fuel Oil #4 Prod. (DEAM mix)	Mercury			4.46e-10			4.46e-10
Manufacturing	Fuel Oil #6 Prod. (DEAM)	Mercury			2.36e-10			2.36e-10
Manufacturing	Japanese Electric Grid	Mercury			3.97e-07			3.97e-07
Manufacturing	LCD CCFL mfg. (CDP)	Mercury				3.99e-06		
Manufacturing	LCD module mfg. (CDP)	Mercury			9.69e-08		1.94e-07	9.69e-08
Manufacturing	LPG Production (DEAM)	Mercury			4.17e-08			4.17e-08
Manufacturing	Natural Gas Prod. (DEAM)	Mercury			2.97e-09			2.97e-09
Manufacturing	US electric grid	Mercury			1.45e-08			1.45e-08
Manufacturing	Fuel Oil #2 Prod. (DEAM)	Mercury compounds			9.27e-14		1.64e-13	9.25e-14

APPENDIX N

	Table N-11. LCD Life-(Uvcle Impact Scores	s trom Me	rcurv-bas	sed Mater	ials		
Life-cycle Stage	Process	Input/Output		•	Impact Scores	by Category		
			Haz. Waste	Solid Waste	ChrTox- pub	ChrTox- occ	Aq. Tox.	Terr. Tox.
Manufacturing	Fuel Oil #4 Prod. (DEAM mix)	Mercury compounds			2.49e-13		4.39e-13	2.48e-13
Manufacturing	Fuel Oil #6 Prod. (DEAM)	Mercury compounds			8.15e-14		1.44e-13	8.14e-14
Manufacturing	LPG Production (DEAM)	Mercury compounds			3.40e-11		6.00e-11	3.39e-11
Manufacturing	Natural Gas Prod. (DEAM)	Mercury compounds			2.26e-14		3.98e-14	2.25e-14
Manufacturing	LCD backlight unit assembly (CDP)	Waste glass, with mercury	7.73e-15					
Use	US electric grid	Mercury			2.80e-06			2.80e-06
End of Life	Fuel Oil #4 Prod. (DEAM mix)	Mercury			-2.05e-09			-2.05e-09
End of Life	LCD incineration (DEAM mix)	Mercury			-8.42e-08			-8.42e-08
End of Life	LPG Production (DEAM)	Mercury			3.42e-12			3.42e-12
End of Life	Natural Gas Prod. (DEAM)	Mercury			-6.23e-10			-6.23e-10
End of Life	US electric grid	Mercury			5.32e-10			5.32e-10
End of Life	Fuel Oil #4 Prod. (DEAM mix)	Mercury compounds			-1.15e-12		-2.02e-12	-1.14e-12
End of Life	LCD incineration (DEAM mix)	Mercury compounds			4.57e-09		8.08e-09	4.57e-09
End of Life	LCD landfilling (CDP)	Mercury compounds			3.96e-09		6.99e-09	3.95e-09
End of Life	LPG Production (DEAM)	Mercury compounds			2.79e-15		4.93e-15	2.78e-15
End of Life	Natural Gas Prod. (DEAM)	Mercury compounds			-4.74e-15		-8.37e-15	-4.73e-15
MP = Materials nrc	Dessing.							

, . ¢ ŭ Ć • Þ Ę 1 2 Ę

Blank = not applicable.

this page intentionally left blank